Knox H. Todd Charles R. Thomas, Jr. Kumar Alagappan *Editors*

Oncologic Emergency Medicine

Principles and Practice Second Edition



Oncologic Emergency Medicine

Knox H. Todd • Charles R. Thomas, Jr. Kumar Alagappan Editors

Oncologic Emergency Medicine

Principles and Practice

Second Edition



Editors Knox H. Todd Former Professor and Founding Chair (Retired) The University of Texas MD Anderson Cancer Center Houston, TX USA

Kumar Alagappan Professor and Chair Department of Emergency Medicine The University of Texas MD Anderson Cancer Center Houston, TX USA Charles R. Thomas, Jr. Professor and Chair Department of Radiation Medicine Knight Cancer Institute Oregon Health & Science University Portland, OR USA

ISBN 978-3-030-67122-8 ISBN 978-3-030-67123-5 (eBook) https://doi.org/10.1007/978-3-030-67123-5

© Springer Nature Switzerland AG 2021, corrected publication 2021

This work is subject to copyright. All rights are reserved by the Publisher, whether the whole or part of the material is concerned, specifically the rights of translation, reprinting, reuse of illustrations, recitation, broadcasting, reproduction on microfilms or in any other physical way, and transmission or information storage and retrieval, electronic adaptation, computer software, or by similar or dissimilar methodology now known or hereafter developed.

The use of general descriptive names, registered names, trademarks, service marks, etc. in this publication does not imply, even in the absence of a specific statement, that such names are exempt from the relevant protective laws and regulations and therefore free for general use.

The publisher, the authors and the editors are safe to assume that the advice and information in this book are believed to be true and accurate at the date of publication. Neither the publisher nor the authors or the editors give a warranty, expressed or implied, with respect to the material contained herein or for any errors or omissions that may have been made. The publisher remains neutral with regard to jurisdictional claims in published maps and institutional affiliations.

This Springer imprint is published by the registered company Springer Nature Switzerland AG The registered company address is: Gewerbestrasse 11, 6330 Cham, Switzerland

This book is dedicated to my wife, Courtenay Carmody, and my two sons, Samuel and Benjamin, whose love and support bring great joy to my life and purpose to my work. In memory of my mother, Mary Kate Todd, who left us much too early. –Knox H. Todd

To my supportive wife, Muriel Elleen; our wonderful two children, Julian Franklin and Aurielle Marie; our parents; and siblings for their love and support of my career path.

In memory of my mother, Ruth Marie Wilson Thomas, who fought gallantly in the war against cancer and whose prayers have blessed me over the past five decades.

-Charles R. Thomas Jr.

This book is dedicated to my wife, Bridget Shields, and my two daughters, Lakshmi and Jyothi.

In memory of my father, Alagappa Alagappan PhD, an international civil servant who traveled to 95 countries. He always felt that learning was a treasure that would follow its owner everywhere.

-Kumar Alagappan

Foreword

This second edition of *Oncologic Emergency Medicine: Principles and Practice* has broadened the scope of emergency care for the oncological patient and rightfully so. More patients are living with cancer than ever before. Many cancer diagnoses are associated with greatly extended lifespans. With greater lifespan, cancer survivors are also at risk of acquiring complications from their past medical, radiation, or surgical care, in addition to an underlying cancer.

Today's patient living with a current or prior cancer diagnosis lives within a network of health care providers. The emergency physician and the oncologist must understand not only the underlying pathophysiology of the disease and its treatment, but also appreciate how patients present in distress and how care is optimized to meet the needs of the patient. Indeed, while using best practice options, the team of health care providers personalizes each patient's treatment, stabilization, and recovery to optimize outcomes, seeking to match the expectations of the patient and family.

In this edition, cancer prevention approaches are discussed. Emergency physicians must be comfortable with cancer prevention approaches to minimize future risk for those living with cancer. As in the first edition, the text provides cutting-edge information related to specific oncologic conditions which may present in the emergency department. At times these conditions may be the initial presentation of the cancer. Other times these conditions may represent complications of therapy, disease progression, and/or secondary infection. Many of these conditions require a time-dependent response to minimize additional morbidity or mortality. This text provides unique information (including discussion of new immunologic therapies) that may be covered only superficially by other books or articles in the field.

The palliative aspects of cancer care are also covered. The learning points covered in these chapters are especially cogent in today's changed world of opioid dependence and COVID-19. Guidance is provided regarding the management of pain, communication about the diagnosis/ progression of cancer, and discussions regarding hospice and other related factors. These represent key components of doctoring which require great skill and understanding of the patient, as well as the disease.

This edition also seeks to extend our understanding of health disparities and vulnerable populations, international approaches to oncological emergencies, and other topics on the cutting-edge cancer research, treatment, and management. The editors are to be congratulated for extending their earlier work and both recognizing and anticipating the needs of care providers in oncologic emergency medicine. They have provided a powerful tool to enhance the care provided to cancer patients.

Jerris R. Hedges, MD, MS, MMM Professor & Dean, Barry & Virginia Weinman – Endowed Chair John A. Burns School of Medicine, University of Hawaii – Manoa Former Interim Director, University of Hawaii Cancer Center Honolulu, HI, USA

Preface

It is with renewed excitement that we present the second edition of *Oncologic Emergency Medicine: Principles and Practice*, the major comprehensive textbook in this emerging field. Against the backdrop of rising numbers of cancer patients and survivors as the US population ages, and of a forecast shortage of cancer care providers, this book is designed to serve as the most authoritative, single-source clinical reference on oncologic emergencies. This comprehensive text was specifically designed to address the complexities of understanding and managing cancer emergencies, with an emphasis on increasing communication and collaboration between emergency physicians and the multiple providers who participate in caring for those with cancer.

The expanded cadre of contributors to the second edition includes a broad spectrum of experts in emergency medicine and nursing, surgical and medical oncology, hematology, diagnostic and interventional radiology, palliative care, psychiatry, critical care, dermatology, oph-thalmology, clinical pharmacy, addiction psychology, social work, and health services research.

Emergency departments account for approximately one-half of all hospital admissions, and this proportion is even greater for those with cancer. While the largest portion of the book focuses on a number of clinical oncologic emergencies and their varied presentations to the emergency department, this text offers the opportunity to address more broadly and systematically the vantage point of emergency physicians who work in a critical hub of patient care: the emergency department. Emergency department visits resulting from disease progression as well as toxicities of anticancer treatments serve as an important patient-oriented metric of cancer care quality. This text emphasizes the critical importance of emergency department care within a comprehensive cancer treatment system. The principles of care will be similar whether the emergency department is in a dedicated clinical cancer care facility or a matrix care structure. The methods of executing best practices may differ based on the structure of the cancer care system; however, the vast majority of emergency care for those with cancer is similar across emergency department settings, whether in academic, community, or hybrid practice. The text is structured to cover multiple fundamental areas of emergency care:

Part I is centered on systems issues in oncologic emergency medicine. We discuss the epidemiology of oncologic emergencies, existing models of emergency department care, informatics, the evolving role of quality measures, patient navigation, and the importance of emergency nursing and social work.

Part II considers the role of emergency medicine in primary and secondary cancer prevention, including smoking cessation, alcohol exposure, ionizing radiation and cervical cancer prevention and detection, as well as screening for lung cancer, colorectal cancer, and melanoma.

Part III will seem perhaps the most familiar to readers and includes a discussion of the evaluation and treatment of a variety of oncologic emergencies, organized by organ systems. This section also covers specific conditions common to many cancer types, including febrile neutropenia, thrombosis, and bleeding.

Part IV examines important issues related to treatment toxicity, including chemotoxicity, radiotoxicity, and post-surgical complications, as well as transplant-related issues and toxicities of novel antineoplastic agents such as checkpoint inhibitors and CAR T-cell therapy.

Part V addresses palliative care issues pertinent to the intersection of emergency medicine and oncology. This section discusses end-of-life care, including the role of palliative surgery, the management of symptoms in those with advanced cancer, approaches to opioid analgesic use (and misuse), and the significance of emergency department use at the end of life.

Part VI deals with contextual issues critical to the subdiscipline of oncologic emergency medicine. It incorporates a chapter on the challenges to emergency medicine and oncology posed by COVID-19 and considerations of ethics, health disparities, and ongoing efforts to advance research and education. Finally, we conclude with a chapter by physicians regarding their personal experiences with cancer.

An added feature to this edition are board review questions accompanied by brief and concise explanations of the answers.

The editors and associate editors are extremely proud of this second edition of *Oncologic Emergency Medicine: Principles and Practice* and we wish to thank all of the contributors who have given their time, insight, and experience to create a truly unique text that will serve as a valuable resource for practitioners, researchers, policy makers, trainees, payors, and administrators, as we care for those with urgent cancer needs. We would like to give a special acknowledgment to Ms. Katherine Kreilkamp of Springer for her efficient and diligent role in textbook development. We are deeply indebted to her and to Springer for their support.

Mendoza, Argentina Portland, OR, USA Houston, TX, USA Knox H. Todd Charles R. Thomas Jr. Kumar Alagappan

Contents

Volume I

Par	t I Systems
1	Epidemiology
2	Models of Care 13 Michael G. Purcell 13
3	Informatics.23James P. Killeen, Christian J. Dameff, Jeremy S. Bass, Brian Khan, Ammar Mandvi, and Nathan Yung23
4	Quality Measures 37 Tracy E. Spinks and Michael G. Purcell
5	Patient Navigation.75David C. Seaberg and Donald Norris II
6	Nursing.83Petra Grami, Sorayah S. Bourenane, Danielle Milling, Karen McFarland, Robert T. Drew, and Linda Koppy
7	Palliative Social Work 109 Robin Rudy Lawson and Alison Snow 109
Par	t II Prevention
8	Tobacco Control
9	Problem Drinking
10	Ionizing Radiation. 139 Richard T. Griffey 139
11	Cervical Cancer Screening
12	Lung Cancer Screening
13	Colorectal Cancer Screening

	٠	٠
v	I	н
^		н

14	Mela Wei-	anoma Screening. 185 Lynn Chong, Thomas M. Pitney, and Michael Sinnott
Par	t III	Evaluation and Treatment
15	Acut Chris	te Airway Obstruction
16	Cent Prash and S	ant Rai, Akhil Shivaprasad, Ivo W. Tremont-Lukats, Sudhakar Tummala
17	Pitui Daria	a Krivosheya and Ian E. McCutcheon
18	<mark>Spin</mark> Jayne	al Cord Compression
19	Head Ray Y	1 and Neck
20	Oph Dian	thalmic Emergencies in Cancer Patients259a Chao, Mathieu F. Bakhoum, and Bita Esmaeli
21	Card Tarel Tawa Ninia Umb Rhia	liovascular Oncologic Emergencies
22	Supe Thor	erior Vena Cava Syndrome
23	Puln Nage	nonary
24	Veno Nich	olas R. Pettit and Jeffrey A. Kline
25	Gast Jessi	roenterology
26	The Linds	Acute Abdomen
27	<mark>Diar</mark> Sai-C	rhea
28	Cons Dem	stipation
29	Endo Sai-C	Description Control of the second secon
30	Nepl Amit	arology
31	<mark>Urol</mark> Shaw	ogy
32	<mark>Gyn</mark> Loise	ecology

33	Orthopedics
34	Dermatology
35	Pediatrics
36	Hyperleukocytosis and Leukostasis
37	Bleeding and Thrombosis
38	Infectious Disease
39	Febrile Neutropenia 513 Christopher J. Coyne and Rahul V. Nene 513
40	Neutropenic Enterocolitis
41	Sickle Cell Disease
42	Radiological and Nuclear Terrorism543Carol J. Iddins, Jason E. Davis, Ronald E. Goans, and Cullen Case Jr.
43	Delirium
44	Suicide
45	Substance Abuse
46	Radiology 589 Jihoon Lim, Keith D. Herr, and Tarek N. Hanna
47	Ultrasound
Vol	ume II

Part IV Toxicities

48	Chemotherapy Toxicities	\$7
49	Treatment Toxicity: Radiation	53
50	Mucositis	19

51	Stoma Complications
52	Hematopoietic Cell Transplant
53	Toxicities of Novel Antineoplastic Therapies 711Eva Rajha and Sai-Ching Jim Yeung
54	Fluorouracil or Capecitabine Overdose
Part	t V Palliative Care
55	Emergency Department Use at End of Life
56	Pain Management
57	Dyspnea in the Dying Patient
58	Cardiopulmonary Resuscitation
59	Palliative Surgery
60	Palliative Care 809 Sangeeta Lamba and Rebecca Goett 809
61	Hospice
62	Communication
63	Palliative Care Research. 843 Paul L. DeSandre and Gretchen E. Bell 843
Part	t VI Contextual
64	Ethics
65	Health Disparities
66	Emergency Oncology in the United Kingdom
67	Cancer Pain Management in Low-Resource Settings
68	Curriculum Development
69	National and Institutional Research Efforts

xiv

70	COVID-19
71	The Physician and Cancer: In Their Own Words
Cor	rection to: Endocrinology and Metabolism
Mu	Itiple-Choice Questions
Ans	wers
Ind	ex

Editors

Knox H. Todd, MD, MPH (Retired) Department of Emergency Medicine, MD Anderson Cancer Center, Houston, TX, USA

Charles R. Thomas Jr., MD Department of Radiation Medicine, Knight Cancer Institute, Oregon Health & Science University, Portland, OR, USA

Kumar Alagappan, MD Department of Emergency Medicine, MD Anderson Cancer Center, Houston, TX, USA

Associate Editors

Patricia A. Brock, MD Department of Emergency Medicine, MD Anderson Cancer Center, Houston, TX, USA

Christopher J. Coyne, MD, MPH Department of Emergency Medicine, Department of Radiation Medicine and Applied Sciences, University of California, San Diego, San Diego, CA, USA

Sarah B. Dubbs, MD Department of Emergency Medicine, University of Maryland, Baltimore, MD, USA

Sai-Ching Jim Yeung, MD, PhD Department of Emergency Medicine, Division of Internal Medicine, The University of Texas MD Anderson Cancer Center, Houston, TX, USA

Contributors

Salahadin Abdi, MD, PhD Department of Pain Medicine, The University of Texas MD Anderson Cancer Center, Houston, TX, USA

Kate Aberger, MD Division of Palliative and Geriatric Medicine, Department of Internal Medicine, St. Joseph's Health, Paterson, NJ, USA

Wissam Abouzgheib, MD Division of Pulmonary and Critical Care, Department of Medicine, Cooper Medical School at Rowan University, Camden, NJ, USA

Zahra Alhajji, MBBS, MD Department of Emergency Medicine, The University of Texas MD Anderson Cancer Center, Houston, TX, USA

Lamees M. Alshaikh, MBBS, MPH Department of Emergency Medicine, Ben Taub General Hospital, Baylor College of Medicine, Houston, TX, USA

Laurie M. Aluce, MD Department of Emergency Medicine, Northwestern Memorial Hospital, Feinberg School of Medicine, Northwestern University, Chicago, IL, USA

Arya Amini, MD Department of Radiation Oncology, City of Hope National Medical Center, Duarte, CA, USA

Brian Badgwell, MD, MS Department of Surgical Oncology, Division of Surgery, MD Anderson Cancer Center, Houston, TX, USA

Ellen S. Baker, MD, MPH Department of Gynecologic Oncology & Reproductive Medicine, The University of Texas MD Anderson Cancer Center, Houston, TX, USA

Mathieu F. Bakhoum, MD, PhD Department of Ophthalmology, University of California, San Diego, La Jolla, CA, USA

Ophthalmic Plastic and Orbital Surgery, Department of Plastic Surgery, MD Anderson Cancer Center, Houston, TX, USA

Gene Bakis, MD Division of Gastroenterology and Hepatology, Department of Medicine, Oregon Health and Science University, Portland, OR, USA

Tarek Barbar, MD Division of Cardiology, Department of Internal Medicine, NewYork-Presbyterian Hospital/Weill Cornell Medical Center, New York, NY, USA

Jeremy S. Bass, MD, MEE Department of Psychiatry, Clinical Informatics, University of California San Diego, San Diego, CA, USA

Gretchen E. Bell, MD Department of Family and Preventive Medicine, Emory University School of Medicine, Atlanta, GA, USA

Janice F. Bell, MN, MPH, PhD Betty Irene Moore School of Nursing, University of California at Davis, Davis, CA, USA

Steven L. Bernstein, MD Department of Emergency Medicine, Yale New Haven Hospital, Yale University School of Medicine, New Haven, CT, USA

Maryann Bishop-Jodoin, MEd Department of Radiation Oncology, University of Massachusetts Medical School, Worcester, MA, USA

Sorayah S. Bourenane, MSN, RN, CEN, CNL Emergency Department & Clinical Decision Unit, The University of Texas MD Anderson Cancer Center, Houston, TX, USA

Tatjana Bozanovic, MD, PhD Clinic for Gynecology and Obstetrics, Clinical Centre of Serbia, University of Belgrade, Belgrade, Serbia

Sherry-Ann Brown, MD, PhD Cardio-Oncology Program, Division of Cardiovascular Medicine, Department of Medicine, Medical College of Wisconsin, Milwaukee, WI, USA

Umberto Campia, MD, MS Division of Cardiovascular Medicine, Department of Medicine, Brigham and Women's Hospital, Harvard Medical School, Boston, MA, USA

Jennifer Carnell, MD Department of Emergency Medicine, Ben Taub General Hospital, Baylor College of Medicine, Houston, TX, USA

Cullen Case Jr., EMPA Radiation Injury Treatment Network, National Marrow Donor Program, Minneapolis, MN, USA

Christopher H. Chang, MD Division of Pulmonary and Critical Care Medicine, Department of Medicine, Oregon Health and Science University, Portland, OR, USA

Diana Chao, MD Intelligent Eye Care, Arcadia, CA, USA

Ophthalmic Plastic and Orbital Surgery, Department of Plastic Surgery, MD Anderson Cancer Center, Houston, TX, USA

Tawanna Charlton, MS Department of Medicine, Cardiology, College of Medicine, University of Florida, Gainesville, FL, USA

Wei-Lynn Chong, MBBS Emergency Department, Princess Alexandra Hospital, Brisbane, QLD, Australia

Victor A. Chow, MD Division of Medical Oncology, University of Washington School of Medicine, Clinical Research Division, Fred Hutchinson Cancer Research Center, Seattle, WA, USA

Robert L. Cloutier, MD, MCR Pediatric Emergency Medicine, Department of Emergency Medicine, Office of Admissions, School of Medicine, Oregon Health & Science University, Portland, OR, USA

Tim Cooksley, MBChB (Hons), FRCPE Department of Acute Medicine, The Christie NHS Foundation Trust, Manchester, Lancashire, UK

Edouard Coupet Jr., MD, MS Department of Emergency Medicine, Yale New Haven Hospital, Yale School of Medicine, New Haven, CT, USA

Angela B. Creditt, DO Department of Emergency Medicine, Virginia Commonwealth University Medical Center (VCU Health), Richmond, VA, USA

Patrick J. Crocker, DO, MS Department of Emergency Medicine, Dell Children's Medical Center, Austin, TX, USA

Department of Emergency Medicine, Brackenridge Hospital, Austin, TX, USA

Edward P. Cutolo Jr., MD Veterans Integrated Service Network 8 (VISN 8), St. Petersburg, FL, USA

Christian J. Dameff, MD, MS Department of Emergency Medicine, Department of Biomedical Informatics, University of California San Diego, San Diego, CA, USA

Shawn Dason, MD Department of Urology, The Ohio State University, Columbus, OH, USA

Jason E. Davis, PhD Oak Ridge Associated Universities, Radiation Emergency Assistance Center/Training Site, Oak Ridge Institute for Science and Education, Oak Ridge, TN, USA

Dany Debs, MD Roger Williams Medical Center, Department of Medicine, Boston University School of Medicine, Providence, RI, USA

Thomas G. DeLoughery, MD Division of Hematology/Medical Oncology, Department of Medicine, Oregon Health & Science University, Portland, OR, USA

Paul L. DeSandre, DO Department of Emergency Medicine, Emory University School of Medicine, Atlanta, GA, USA

Burton F. Dickey, MD Department of Pulmonary Medicine, The University of Texas MD Anderson Cancer Center, Houston, TX, USA

Eric W. Dickson, MD, MHCM Department of Emergency Medicine, UMass Memorial Health Care, University of Massachusetts Medical School, Worcester, MA, USA

Michelle Dimza, DO Division of General Internal Medicine, Department of Medicine, Cardiology, University of Florida, Gainesville, FL, USA

Stephen Dobbin, MBChB, MRCP Institute of Cardiovascular and Medical Sciences, University of Glasgow, Scotland, UK

Danielle J. Doberman, MD, MPH, HMDC Section of Palliative Medicine, Department of Emergency Medicine, Johns Hopkins Hospital, Johns Hopkins University, Baltimore, MD, USA

Robert T. Drew, MBA, MSN, RN, CEN, CCRN, NE-BC Emergency Department & Clinical Decision Unit, The University of Texas MD Anderson Cancer Center, Houston, TX, USA

Leslie R. Ellis, MD, MSHPEd Division of Hematology and Oncology, Department of Internal Medicine, Comprehensive Cancer Center, Wake Forest Baptist Health, Winston-Salem, NC, USA

Ahmed Elsayem, MD Department of Emergency Medicine, The University of Texas MD Anderson Cancer Center, Houston, TX, USA

Brintha K. Enestvedt, MD, MBA Division of Gastroenterology and Hepatology, Department of Medicine, Oregon Health and Science University, Portland, OR, USA

Bita Esmaeli, MD Ophthalmic Plastic and Orbital Surgery, Department of Plastic Surgery, MD Anderson Cancer Center, Houston, TX, USA

Maria A. Farberov, MPA Clinical Research Education, Department of Medicine, Northwell Health, Staten Island University Hospital, Staten Island, NY, USA

Linda Ferrari, MD Colorectal Department, Guy's and St. Thomas' NHS Foundation Trust, London, UK

Alessandro Fichera, MD Division of Colon and Rectal Surgery, Department of Surgery, Baylor University Medical Center, Dallas, TX, USA

Thomas J. FitzGerald, MD Department of Radiation Oncology, UMass Memorial Health Care, University of Massachusetts Medical School, Worcester, MA, USA

Michael G. Fradley, MD Division of Cardiovascular Medicine, Department of Medicine, Perelman Center for Advanced Medicine, Perelman School of Medicine at the University of Pennsylvania, Philadelphia, PA, USA

Clifton D. Fuller, MD, PhD Department of Radiation Oncology, The University of Texas MD Anderson Cancer Center, Houston, TX, USA

Arjun K. Ghosh, MBBS, MRCP, MSc, PhD Barts Heart Centre, Barts NHS Health Trust, Department of Cardiology, St. Bartholomew's Hospital, London, England, UK

Ronald E. Goans, PhD, MD, MPH MJW Corporation, Clinton, TN, USA

Rebecca Goett, MD Department of Emergency Medicine, Rutgers New Jersey Medical School, Newark, NJ, USA

Petra Grami, DNP, RN, CCRN, NE-BC, CVRN Critical Care, Clinical Decision Unit and Acute Dialysis, Nursing Administration, MD Anderson Cancer Center, Houston, TX, USA

Richard T. Griffey, MD, MPH Department of Emergency Medicine, Department of Medicine, Barnes Jewish Hospital/Washington University School of Medicine, St. Louis, MO, USA

Neil D. Gross, MD Division of Surgery, Department of Head and Neck Surgery, The University of MD Anderson Cancer Center, Houston, TX, USA

Daniel C. Han, MD Department of Radiation Oncology, UMass Memorial Health Care, University of Massachusetts Medical School, Worcester, MA, USA

Tarek N. Hanna, MD Department of Radiology and Imaging Sciences and Department of Emergency Medicine, Emory University, Atlanta, GA, USA

Joanna Harp, MD Department of Dermatology, Weill Cornell Medical College, New York Presbyterian Hospital, New York, NY, USA

Mohammad Adrian Hasdianda, MD, MSc, MMS Department of Emergency Medicine, Brigham and Women's Hospital, Boston, MA, USA

Keith D. Herr, MD Department of Radiology and Imaging Sciences and Department of Emergency Medicine, Emory University, Atlanta, GA, USA

Caitlin Hodge, MD, MPH Department of Surgery, Abington Hospital-Jefferson Health, Abington, PA, USA

Evelyn Horn, MD Department of Medicine, Cardiology Division, Perkin Center for Heart Failure, New York-Presbyterian Hospital/Weill Cornell Medical Center, New York, NY, USA

Christopher O. Hoyte, MD Rocky Mountain Poison and Drug Safety, Denver Health and Hospital Authority, Denver, CO, USA

Department of Emergency Medicine, University of Colorado School of Medicine, Aurora, CO, USA

Kami M. Hu, MD Department of Emergency Medicine, Department of Internal Medicine, University of Maryland School of Medicine, Baltimore, MD, USA

Carol J. Iddins, MD Radiation Emergency Assistance Center/Training Site, Oak Ridge Institute for Science and Education, Oak Ridge, TN, USA

Kenneth V. Iserson, MD, MBA Department of Emergency Medicine, University of Arizona, Tucson, AZ, USA

Tami N. Johnson, PharmD, BCPS, FCCM Department of Pharmacy, The University of Texas MD Anderson Cancer Center, Houston, TX, USA

Jenna M. Kahn, MD Department of Radiation Medicine, Oregon Health & Science University, Portland, OR, USA

Konrad Karasek, MD Department of Emergency Medicine, Ben Taub General Hospital, Baylor College of Medicine, Houston, TX, USA

Brian Khan, MD, MBA Department of Medicine, Clinical Informatics, University of California San Diego, San Diego, CA, USA

James P. Killeen, MD Department of Emergency Medicine, Department of Biomedical Informatics, University of California San Diego, San Diego, CA, USA

Michael J. Kim, MD Division of Emergency Medicine, Department of Medicine, Barnes Jewish Hospital/Washington University School of Medicine, St. Louis, MO, USA

Jeffrey A. Kline, MD Department of Emergency Medicine, Eskenazi and Methodist Hospitals, Indiana University School of Medicine, Indianapolis, IN, USA

Eric C. Ko, MD, PhD Department of Radiation Oncology, UMass Memorial Health Care, University of Massachusetts Medical School, Worcester, MA, USA

Linda Koppy, BSN, BBA Emergency Center, The University of Texas MD Anderson Cancer Center, Houston, TX, USA

Tatiana Starr Kreitzer, MA University of South Florida College of Nursing, Tampa, FL, USA

Daria Krivosheya, MD Department of Neurological Surgery, Cleveland Clinic Lerner College of Medicine, Akron, OH, USA

Amit Lahoti, MD Department of Nephrology, The University of Texas MD Anderson Cancer Center, Houston, TX, USA

Sangeeta Lamba, MD Department of Emergency Medicine, Rutgers New Jersey Medical School, Newark, NJ, USA

Ninian N. Lang, BSc (Hons) MBChB, PhD, FRCP Institute of Cardiovascular and Medical Sciences, University of Glasgow, Scotland, UK

Rebecca S. Lash, PhD, MPP, RN School of Nursing, Indiana University, Fort Wayne, IN, USA

Fran Laurie, BS Department of Radiation Oncology, University of Massachusetts Medical School, Worcester, MA, USA

Robin Rudy Lawson, LMSW Empath Health, Suncoast Hospice, St. Petersburg, FL, USA

Cheryl T. Lee, MD Department of Urology, The Ohio State University, Columbus, OH, USA

Stephen Y. Liang, MD, MPHS Divisions of Emergency Medicine and Infectious Diseases, Department of Medicine, Washington University School of Medicine, St. Louis, MO, USA

Jihoon Lim, MD Department of Radiology and Imaging Sciences and Department of Emergency Medicine, Emory University, Atlanta, GA, USA

Patrick P. Lin, MD Department of Orthopedic Oncology, The University of Texas MD Anderson Cancer Center, Houston, TX, USA

Demis N. Lipe, MD Department of Emergency Medicine, Division of Internal Medicine, The University of Texas MD Anderson Cancer Center, Houston, TX, USA

Bernard L. Lopez, MD, MS Department of Emergency Medicine, Sidney Kimmel Medical College, Thomas Jefferson University & Hospitals, Philadelphia, PA, USA

Maggie Q. Ma, PharmD, BCPS Department of Pharmacy, The University of Texas MD Anderson Cancer Center, Houston, TX, USA

Nagendra Y. Madisi, MD Division of Pulmonary and Critical Care, Department of Medicine, Cooper Medical School at Rowan University, Camden, NJ, USA

Syed S. Mahmood, MD, MPH Division of Cardiology, Department of Internal Medicine, New York-Presbyterian Hospital/Weill Cornell Medical Center, New York, NY, USA

Ammar Mandvi, MD Department of Family Medicine, Clinical Informatics, University of California San Diego, San Diego, CA, USA

David E. Manthey, MD Department of Emergency Medicine, Wake Forest Baptist Health, Winston-Salem, NC, USA

Jennifer L. Maranki, MD, MSc Division of Gastroenterology and Hepatology, Department of Medicine, Penn State Hershey Medical Center, Penn State College of Medicine, Hershey, PA, USA

Alina Markova, MD Department of Dermatology, Memorial Sloan Kettering Cancer Center, New York, NY, USA

Bonnie K. Marr, MD Section of Palliative Medicine, Johns Hopkins Hospital, Baltimore, MD, USA

Danielle M. McCarthy, MD, MS Department of Emergency Medicine, Northwestern Memorial Hospital, Feinberg School of Medicine, Northwestern University, Chicago, IL, USA

Ian E. McCutcheon, MD Department of Neurosurgery, The University of Texas MD Anderson Cancer Center, Houston, TX, USA

Karen McFarland, MSN, BSN, VA-BC Emergency Department & Clinical Decision Unit, The University of Texas MD Anderson Cancer Center, Houston, TX, USA **Steven McGaughey, MD** Department of Emergency Medicine, Oregon Health & Science University, Portland, OR, USA

Kayla McMillan, BPH Department of Oncology, Juravinski Cancer Centre, Hamilton, ON, Canada

Danielle Milling, MSN, RN, CPN, CNL Emergency Center and Clinical Decision Unit, The University of Texas MD Anderson Cancer Center, Houston, TX, USA

Joel Moll, MD Department of Emergency Medicine, Virginia Commonwealth University, Richmond, VA, USA

Marshall T. Morgan, MD Division of Emergency Medicine, Department of Medicine, UCLA Medical Center, Los Angeles, CA, USA

Lakshmi Mudambi, MBBS Division of Pulmonary and Critical Care Medicine, Department of Medicine, Oregon Health and Science University, Portland, OR, USA

Division of Pulmonary and Critical Care, VA Portland Health Care System, Portland, OR, USA

Alexandra Murphy, MBBS Department of Cardiology, Austin Health & The University of Melbourne, Melbourne, VIC, Australia

Tarek Nafee, MD, MMSc Roger Williams Medical Center, Department of Medicine, Boston University School of Medicine, Providence, RI, USA

Rahul V. Nene, MD, PhD Department of Emergency Medicine, University of California, San Diego, San Diego, CA, USA

Natalie R. Neumann, MD Department of Emergency Medicine, Yale University School of Medicine, New Haven, CT, USA

Nathanial S. Nolan, MD, MPH Division of Infectious Disease, Department of Medicine, Barnes Jewish Hospital/Washington University School of Medicine, St. Louis, MO, USA

Donald Norris II, MD Department of Emergency Medicine, Summa Health, Akron, OH, USA Department of Emergency Medicine, Northeast Ohio Medical University, Rootstown, OH, USA

Jennifer A. Novak, MD, MS Department of Radiation Oncology, City of Hope National Medical Center, Duarte, CA, USA

Tochukwu M. Okwuosa, DO Division of Cardiology, Department of Internal Medicine, Rush University Medical Center, Chicago, IL, USA

Kei Ouchi, MD, MPH Department of Emergency Medicine, Brigham and Women's Hospital, Boston, MA, USA

Ki Park, MD, MS UF Women's Cardiovascular Health Sciences, Malcom Randall VAMC, Division of Cardiovascular Medicine, Department of Medicine, Cardiology, University of Florida, Gainesville, FL, USA

Steven D. Passik, PhD Scientific Affairs, Research and Development, Collegium Pharmaceutical, Stoughton, MA, USA

Tanja Pejovic, MD, PhD Department of Obstetrics and Gynecology, Oregon Health & Science University, Portland, OR, USA

Marcia A. Perry, MD Department of Emergency Medicine, University of Michigan, Ann Arbor, MI, USA

Nicholas R. Pettit, DO, PhD Department of Emergency Medicine, Eskenazi Hospital, Indiana University School of Medicine, Indianapolis, IN, USA

Natacha Phoolcharoen, MD Department of Gynecologic Oncology & Reproductive Medicine, The University of Texas MD Anderson Cancer Center, Houston, TX, USA

Department of Obstetrics and Gynecology, King Chulalongkorn Memorial Hospital, Faculty of Medicine, Chulalongkorn University, Bangkok, Thailand

Lindsey Picard, MD Department of Emergency Medicine, Strong Memorial Hospital, University of Rochester Medical Center, Rochester, NY, USA

Thomas M. Pitney, MBBS Department of Dermatology, The Mater Public Hospital, Brisbane, QLD, Australia

Maria Isabel Camara Planek, MD Department of Medicine, Rush University Medical Center, Chicago, IL, USA

Raaj K. Popli, MD Department of Internal Medicine, University of Central Florida College of Medicine, Orlando, FL, USA

Michael G. Purcell, MD Department of Emergency Medicine, Ohio State University Wexner Medical Center, Columbus, OH, USA

Prashant Rai, MBBS Department of Neurology, University of Texas Medical Branch, Galveston, TX, USA

Eva Rajha, MD Department of Emergency Medicine, Baylor College of Medicine, Houston, TX, USA

Anis Rashid, MD Department of Psychiatry, The University of Texas MD Anderson Cancer Center, Houston, TX, USA

Paul S. Rava, MD, PhD Department of Radiation Oncology, UMass Memorial Health Care, University of Massachusetts Medical School, Worcester, MA, USA

Sarah C. Reed, PhD, MPH, MSW Division of Social Work, California State University, Sacramento, CA, USA

Jenny L. Ren, BS, BA Department of Emergency Medicine, Division of Internal Medicine, The University of Texas MD Anderson Cancer Center, Houston, TX, USA

Cielito Reyes-Gibby, DrPH Department of Emergency Medicine, Division of Internal Medicine, Department of Biostatistics, The University of Texas MD Anderson Cancer Center, Houston, TX, USA

Adam Rzetelny, PhD Department of Medical Affairs, Collegium Pharmaceuticals, Stoughton, MA, USA

Allison Sacher, MD Department of Radiation Oncology, UMass Memorial Health Care, University of Massachusetts Medical School, Worcester, MA, USA

Novneet N. Sahu, MD, MPA Department of Emergency Medicine, Rutgers New Jersey Medical School, Newark, NJ, USA

Mila Pontremoli Salcedo, MD, PhD Department of Gynecologic Oncology & Reproductive Medicine, The University of Texas MD Anderson Cancer Center, Houston, TX, USA

The Obstetrics and Gynecology Department, Federal University of Health Sciences of Porto Alegre (UFCSPA)/Santa Casa de Misericordia Hospital of Porto Alegre, Porto Alegre, Brazil

Marcelo Sandoval, MD Department of Emergency Medicine, Division of Internal Medicine, The University of Texas MD Anderson Cancer Center, Houston, TX, USA

David C. Seaberg, MD Department of Emergency Medicine, Summa Health, Akron, OH, USA Department of Emergency Medicine, Northeast Ohio Medical University, Rootstown, OH, USA

Hsien Seow, PhD Department of Oncology, McMaster University, Hamilton, ON, Canada

Ahmad Shabsigh, MD Department of Urology, The Ohio State University, Columbus, OH, USA

Akhil Shivaprasad, BE, MBBS Stanley H. Appel Department of Neurology, Houston Methodist Hospital, Houston, TX, USA

Ashley Shreves, MD Palliative Medicine & Supportive Care, Emergency Medicine, Ochsner Medical Center, New Orleans, LA, USA

Veronica Sikka, MD, PhD, MHA, MPH Clinical Contact Center, VA Sunshine Healthcare Network (VISN 8), St. Petersburg, FL, USA

Associate Professor of Emergency Medicine, School of Medicine, University of Central Florida, Orlando, FL, USA

Kevin Sing, MD Division of Hematology, Oncology, and Palliative Care, Department of Medicine, Virginia Commonwealth University Medical Center (VCU Health), Richmond, VA, USA

Michael Sinnott, MBBS, FACEM, FRACP The University of Queensland Medical School, Brisbane, QLD, Australia

Alison Snow, PhD, LCSW-R, OSW-C Cancer Supportive Services, Cancer Center, Mount Sinai Beth Israel, New York, NY, USA

Sherwin Soltani, MD Department of Emergency Medicine, Ben Taub General Hospital, Baylor College of Medicine, Houston, TX, USA

Eugene L. Son, MD Department of Head & Neck Surgery, Kaiser Permanente, Fontana, CA, USA

Tracy E. Spinks, BBA Medical Clinical Operations, Oncology Value Creation Team, UnitedHealthcare, Sugarland, TX, USA

Kathleen M. Stephanos, MD Department of Emergency Medicine, Strong Memorial Hospital, University of Rochester Medical Center, Rochester, NY, USA

Harsh P. Sule, MD, MPP Department of Emergency Medicine, Rutgers New Jersey Medical School, Newark, NJ, USA

Katy M. Toale, PharmD, BCPS Department of Pharmacy, The University of Texas MD Anderson Cancer Center, Houston, TX, USA

Rhian M. Touyz, MBBCh, PhD Institute of Cardiovascular and Medical Sciences, BHF Glasgow Cardiovascular Research Centre, University of Glasgow, Glasgow, Scotland, UK

Ivo W. Tremont-Lukats, MD Kenneth R. Peak Brain and Pituitary Tumor Center, Department of Neurosurgery, Houston Methodist Hospital, Houston, TX, USA

Sherilyn A. Tuazon, MD, MS Division of Medical Oncology, University of Washington School of Medicine, Clinical Research Division, Fred Hutchinson Cancer Research Center, Seattle, WA, USA

Sudhakar Tummala, MD Department of Neuro-Oncology, MD Anderson Cancer Center, Houston, TX, USA

Sara Tyebally, MBBS, BSc Barts Heart Centre, Barts NHS Health Trust, Department of Cardiology, St. Bartholomew's Hospital, London, England, UK

Masumi Ueda, MD Division of Medical Oncology, University of Washington School of Medicine, Clinical Research Division, Fred Hutchinson Cancer Research Center, Seattle, WA, USA **Federico E. Vaca, MD, MPH** Department of Emergency Medicine, Yale New Haven Hospital, Yale School of Medicine, New Haven, CT, USA

Jayne M. Viets-Upchurch, MD Department of Emergency Medicine, Division of Internal Medicine, The University of Texas MD Anderson Cancer Center, Houston, TX, USA

Phuong T. Vo, MD Division of Medical Oncology, University of Washington School of Medicine, Clinical Research Division, Fred Hutchinson Cancer Research Center, Seattle, WA, USA

Gregory A. Volturo, MD Department of Emergency Medicine, UMass Memorial Health Care, University of Massachusetts Medical School, Worcester, MA, USA

Ngoc H. Vu, PharmD, BCPS Department of Pharmacy, The University of Texas MD Anderson Cancer Center, Houston, TX, USA

Loise W. Wairiri, MBBS Department of Radiation Medicine, Oregon Health & Science University, Portland, OR, USA

Ray Y. Wang, MD Department of Otolaryngology-Head and Neck Surgery, Baylor College of Medicine, Houston, TX, USA

Robert C. Weinschenk, MD Department of Orthopedic Oncology, The University of Texas MD Anderson Cancer Center, Houston, TX, USA

Thomas C. Williamson, MD Division of Hematology and Oncology, Department of Internal Medicine, Comprehensive Cancer Center, Wake Forest Baptist Health, Winston-Salem, NC, USA

Samantha J. Yarmis, MD Department of Emergency Medicine, Department of Internal Medicine, University of Maryland Medical Center, Baltimore, MD, USA

Shiraz Yazdani, MD Interventional Pain Management, Lubbock Spine Institute, Lubbock, TX, USA

Jessica X. Yu, MD, MS Division of Gastroenterology and Hepatology, Department of Medicine, Oregon Health and Science University, Portland, OR, USA

Sharleen Yuan, MA, PhD, MD Department of Emergency Medicine, Department of Internal Medicine, University of Maryland Medical Center, Baltimore, MD, USA

Nathan Yung, MD, MS Department of Medicine, Clinical Informatics, University of California San Diego, San Diego, CA, USA

Ideen Zeinali, MD Department of Emergency Medicine, Ben Taub General Hospital, Baylor College of Medicine, Houston, TX, USA

Allison Zimmerman, MS, MD Department of Medicine, Rush University Medical Center, Chicago, IL, USA

Part I

Systems

Epidemiology

Rebecca S. Lash, Janice F. Bell, and Sarah C. Reed

Overview

Improving the quality of cancer care and reducing preventable health system use are goals of increasing importance to health practitioners and policy makers in the United States (US) and internationally [1–3]. In 2020, it is estimated that over 1.8 million new cases of cancer will be detected in the United States, over 600,000 deaths will result from cancer, and the estimated annual cost of cancer care is expected to reach to more than \$170 billion [4, 5]. Globally, the incidence of cancer is expected to increase more than 60%, to nearly 30 million, between 2020 and 2040 [6]. Furthermore, as global cancer incidence increases, the burden of cancer care is expected to fall increasingly on low-income countries [7]. This will have significant impacts for the healthcare systems in all countries as they address access to primary and specialty for a growing number of oncologic patients.

In the United States, emergency departments (EDs) serve as a significant source of urgent and safety-net care. In 2014, there were 137.8 million ED visits, reflecting an increase of nearly 15% compared to 2006 [8]. Over the last decade, there has been increased scrutiny of EDs as a source of potentially preventable care because of the high cost of care, potential delays in care, and crowding concerns. As policy makers and practitioners strive to improve quality and reduce fragmentation of cancer care, reducing ED visits is frequently a goal of care coordination program and cost reduction effort. There is a particular interest in reducing visits for concerns labeled as "avoidable" or "preventable"; however, no consensus definitions exist to define or identify such visits [9–12].

R. S. Lash (🖂)

School of Nursing, Indiana University, Fort Wayne, IN, USA

J. F. Bell

Betty Irene Moore School of Nursing, University of California at Davis, Davis, CA, USA

S. C. Reed Division of Social Work, California State University, Sacramento, CA, USA Approximately 4% of all adult ED visits in the United States are for cancer-related complaints [13]. Reports from Australia, the United Kingdom (UK), Brazil, and South Korea highlight concerns about the growing number of cancer patients and the increasing burden of care on EDs in the management of unscheduled care [9, 14–16]. A 2017 National Health Service (NHS) report emphasizes similar patterns in the United Kingdom with dramatic increases in ED presentations related to cancer and concomitantly high rates of inpatient admissions—often associated with poor patient experience, poor coordination of care, poor communication, and fragmented patient care pathways [17].

Oncologic patients present for care in the ED at all points across the cancer care continuum from diagnosis through treatment, survivorship, and end-of-life (Fig. 1.1) [1]. ED visits in this population can range from medical emergencies for those undergoing cancer treatment to nonurgent administrative decisions such as sending a patient to the emergency department for hospital admission. Symptoms—related to cancer or its treatment—may play a role at all points across the continuum; similarly, oncologic patients may present to the ED for concerns or events that are completely unrelated to their cancer (e.g., motor vehicle collisions, musculoskeletal injuries, lacerations, etc.). Importantly, oncologic patients may incur multiple visits for the same issue or for different reasons.

ED use for oncologic patients reflects a complex interaction of individual and contextual factors—including provider behavior, health system characteristics, and health policies. This complexity is well characterized by Chen et al., who show interrelated causal loops to describe a broad array of factors that influence ED use in this population—individual factors (e.g., access to care/insurance, ability to use services), provider factors (e.g., knowledge, skills, communication, referrals, and access to specialists), health system factors (e.g., bed capacity), and policy factors (e.g., availability of social care) (Fig. 1.2) [18]. While these causal loops are identified based on research derived from the United Kingdom, in the context of a national healthcare system,

© Springer Nature Switzerland AG 2021



K. H. Todd et al. (eds.), Oncologic Emergency Medicine, https://doi.org/10.1007/978-3-030-67123-5_1



Fig. 1.1 Emergency department visits across the cancer care trajectory, modeled on the Institute of Medicine's continuum of cancer care [1]



Fig. 1.2 Causal loop diagram of factors that influence ED use by oncologic patients. (From Chen, et al. [18] with permission CC by 4.0)

similar factors apply to ED use across settings. In the United States, access to care is further influenced by insurance status and an individual's ability to pay for care—factors which hold less influence in settings with nationalized healthcare systems.

Early investigations of the ED role in cancer care focused on cancers that were diagnosed in the ED and therefore represent a failure of outpatient detection and potential treatment delays [19, 20]. While these factors remain of concern, more recent studies have moved to enumerate visits in cohorts of patients, describe the type of care provided, and identify potentially preventable ED use [10, 13, 21-23]. Despite these efforts, the epidemiology of ED use by oncologic patients, including patterns and determinants of use, remains poorly understood. Few studies describe the incidence of ED use by cancer patients across cancer types, in population-based samples, and even fewer explore the burden of cancer visits experienced by patients or by ED providers [11]. Numerous studies examine complications, hospital admission, and/or readmission rates for specific cancers, treatment regimens, or procedures, but few of these examine how readmission or complications impact emergency department use [24–31].

In light of the importance of ED visits in the care of oncologic patients, this chapter reviews what is known about ED use among oncologic patients, specifically:

- Distribution (i.e., frequency, incidence, and disposition) of visits
- Determinants of use
- Preventability of visits

Distribution of ED Visits Among Oncologic Patients

Data Sources

In the United States, data sources capturing cancer type, cancer stage, treatment, and health system utilization are not robust. For example, administrative datasets capturing healthcare utilization data may not include cancer diagnosis, treatment information, or complete comorbidity data (e.g., Medical Expenditure Panel Survey (MEPS) [32], Nationwide Emergency Department Sample (NEDS) [33], National Hospital Ambulatory Medical Care Survey (NHAMCS) [34]). Likewise, cancer registries collect robust specific cancer and initial treatment information; however, they may not capture full health services utilization data, including ED visits [35]. The Surveillance, Epidemiology, and End Results (SEER) program includes population-based information about cancer incidence and survival but does not provide

related health services data [36]. While researchers may link available datasets to reduce these gaps in information, limitations in diagnosis, treatment, or health service use remain. Therefore, the frequency and incidence of cancer patients visiting EDs in the United States is difficult to ascertain, particularly for large population-based samples [11].

Studies of the incidence and frequency of ED use by cancer patients are based on either the visit-level or patient-level unit of analysis. Visit-level analyses, for example, provide information about the quantity of care provided to cancer patients by EDs and indicate which diagnoses are associated with the visits. Visit-level analyses may not account for multiple visits made by the same patient, and these datasets may lack patient-specific information, such as diagnosis and treatment data. Patient-level analyses, in contrast, can identify characteristics of patients associated with ED use, thus providing more insight into determinants of health system use.

Estimates

Cancer-Related ED Visits In the United States, it is estimated that over 4% of all ED visits are for cancer-related reasons [13, 21]. Geographically, patterns of ED use are consistent across regions of the country (Northeast, Midwest, South, West) [21]. The number of ED visits varies by primary cancer diagnosis. The top cancers associated with ED visits include lung, 10–27% [13, 22, 23, 37, 38]; breast, 6–15% [13, 22, 23, 37, 38]; colon, 6–12% [13, 22, 23, 37, 38]; prostate, 5–11% [13, 22, 23, 37, 38]; multiple cancers, 10% [13]; and female reproductive or genital, 6–7% [13, 22, 23].

Incidence of ED Use Among Oncologic Patients Among all oncologic patients, the presence of any ED visit and the distribution and timing of ED visits are not well established, in part, because estimates depend on a specific precipitating potential triggering event under study. For instance, ED visits may be measured within a particular time frame (e.g., from diagnosis or from treatment, such as surgery, radiation, or chemotherapy), and the time frames vary greatly in the literature [11]. For example, one study of breast cancer patients receiving a mastectomy reported that 3% of the sample had an ED visit within 30 days of surgery [39], while 11% of high-risk patients with head and neck cancer receiving radiation had an ED visit during treatment or within 90 days of treatment completion [40].

Using a standardized 30-day visit rate, studies that examined postsurgical periods reported 2-12% of the sample visited an ED within 30 days after surgery, and a study that evaluated a post-chemotherapy time frame demonstrated 5% of the sample visited the ED [11]. Estimates provided by the

few available population-based studies tend to report higher ED use than those that focus on smaller within-setting study samples [11].

Time from diagnosis offers a consistent measure to examine ED use by oncologic patients across cancer types, although it is important to consider the factors that impact time to diagnosis. As expected, based on differences in diagnostic and treatment patterns, the incidence of ED visits varies by cancer type. Estimates of ED use by cancer type and time from diagnosis at 30, 180, and 365 days from diagnosis are provided in Table 1.1 [23, 41–43].

Multiple Visits A large proportion of cancer patients have multiple ED visits. Between 2009 and 2010, among patients with all cancer types in California, 20% had one ED visit, 8% had two visits, and 7% had three or more visits within 180 days of diagnosis. Among those with at least one ED visit, 44% had two or more visits and 21% had three or more visits [23]. In a national sample of colon cancer patients who visited the ED (55%), 24% had two visits and 25% had three

 Table 1.1
 Cumulative percentage of cancer patients with at least one

 ED visit by time from diagnosis^a

	Time fro	rom diagnosis	
Cancer type	30 days	180 day	365 days
All	17	35, 44–69 [41]	44
Bladder	21	44	54
Brain	39	60	68
Breast	5	22	31, 15–21 [42]
Colon	20	41	49, 55 [43]
Digestive	26	54	63
Endocrine	7	19	25
Eye	6	18	26
Gynecological	17	36	44
Hodgkin lymphoma	18	43	46
Ill-defined/unknown	36	53	57
Leukemia	26	45	53
Liver	29	54	63
Lung	30	55	64
Male genital (non-prostate)	16	28	36
Melanoma	5	14	22
Myeloma	28	53	63
Non-Hodgkin lymphoma	22	44	51
Oral	12	39	48
Other	20	42	53
Pancreas	37	62	69
Prostate	6	17	25
Respiratory (non-lung)	18	43	52
Stomach	27	55	63
Urinary	21	39	47

^aThese estimates are from four population-based studies that provide data on ED visits by cancer patients and time from diagnosis. Unless specified, data are derived from California's state-based data from 2009 to 2010 [23] or more visits within 1 year of diagnosis; importantly, those with three or more visits (14% of the sample) accounted for over half the total number of visits [43]. Not surprisingly, these rates of multiple visits are substantially higher than in the general US population where 6.5% have two or more visits to an ED annually [44].

Disposition While inpatient admission rates vary by cancer type, patients with cancer who visit the ED tend to have higher rates of admission (49–63%) and are more than twice as likely to be admitted than non-cancer patients [13, 21, 23, 37, 45].

International Perspective on the Incidence and Frequency of ED Use ED use by cancer patients outside the United States is also characterized by high frequency and incidence. For example, during a 1-year study period at a hospital in Japan, 8% of individuals who visited the ED had a cancer diagnosis [46]. Reports from Australia indicate that 40% of cancer patients visited the ED at least once in the year following diagnosis, and 2.4% of all ED visits were made by cancer patients [47, 48]. One single-site study from Australia reported 38% of breast cancer patients receiving chemotherapy presented to the ED within 30 days of chemotherapy administration [49].

Similarly, over 53% of women in Ontario, Canada, who completed at least one cycle of chemotherapy for breast cancer had an ED visit within 30 days of treatment [50]. Among Canadian women over the age of 65 undergoing surgery for breast cancer, 13% had an ED visit within 45 days of surgery [51]. Another Canadian study examining ED use for head and neck cancer patients undergoing surgery found 8.4% visited in ED within 30 days of surgery [52]. These studies demonstrate that despite an indication of high ED utilization by oncologic patients internationally, similar methodological issues persist, and direct comparison of study results across countries is complicated by a focus on different cancers, conditions, treatments, contexts, and time frames of ED use.

Likewise, studies outside the United States report high frequencies of patients with multiple ED visits. In Taiwan, 12% of cancer patients who visited an ED had multiple visits in a 1-year study period, and for those specifically with head and neck cancer, nearly 70% had more than one ED visit over a 12-year study period [53, 54]. A report from the Bureau of Health Information in New South Wales, Australia, indicates that 10% of cancer patients make three or more visits in the year following diagnosis [48]. A separate Australian study found that up to 63% of cancer patients visiting the ED had multiple visits [47]. Among Canadian breast cancer patients receiving chemotherapy, nearly 37% of those within an ED visits within 30 days of treatment had multiple visits [50].

Determinants of ED Use Among Cancer Patients

The Andersen Behavioral Model of Health Services Use provides an important framework to understand the determinants of ED use among cancer patients [55]. Specifically, in this model, health service use is determined by the complex interaction between predisposing characteristics, enabling resources, and need factors. Predisposing characteristics include demographic factors, individuals' beliefs, and aspects of social structure which are not intended to directly explain but rather to help understand differences in the use of health services [55]. Enabling resources include individual and family factors (e.g., income or health insurance) and community characteristics (e.g., region of residence). Need or illness-level factors include both perceived and evaluated health status including symptoms and diagnoses.

The predisposing, enabling, and need factors associated with ED use and frequent ED use tend to be similar across the literature-whether in studies of specific chronic conditions such as diabetes or those limited to subpopulations such as older adults [56, 57]. In the US general population, predisposing factors associated with higher ED visit rates include age and sex, with higher rates among adults age 65 years and older age and among females compared to males; rates are lower among those living in the West compared to other regions [8]. Enabling factors associated with higher rates of ED use include private insurance and Medicaid compared to Medicare or no insurance and residence in low-income areas [8]. In terms of need, a medical diagnosis-including abdominal pain, chest pain, back problems, urinary tract infections, or skin infections-is most often listed as the primary concern for an ED visit [8].

Identifying predisposing, enabling, and need factors for ED visits among oncologic patients presents challenges due to the previously discussed methodological issues including differences in study populations (e.g., by cancer type, country of residence), treatment, the complications of interest, and time frame studied. Among US studies, there is great heterogeneity in the time period examined (e.g., in the first year after diagnosis vs. within 30 days of treatment), the number of participants (n = 220 to 89,311), and diagnoses included [11, 43]. Some studies focus on older adults precluding the ability to examine age as a determinant of ED use. Many of the available studies focus solely on one or more of the top four most prevalent cancers (i.e., prostate, breast, lung, and colorectal).

Predisposing Factors

Significant predisposing factors for oncologic patients of increased ED include race, age, and gender. Specifically, nonwhite [43] and African American [56, 58] compared with white, non-Hispanic race/ethnicity and older age compared to younger age [43, 56] were associated with more ED use. Additionally, being male compared to female sex resulted in increased rates of ED use [59, 60].

Enabling Factors

As in the general population, enabling factors associated with increased ED use for oncologic patients are area of residence and income level in the community of residence, insurance status, and marital status. Those with urban and metropolitan compared to rural residence had higher levels of ED use [56]. Being eligible for Medicaid due to blindness or disability [61], or residing in a state in which Medicaid copayments increased and prescription drug and other benefits were reduced [41], was associated with more ED use. Being unmarried compared to married [56] and residing in a census tract with low median income [56] were also associated with increased ED use.

Need Factors

Multiple conceptualizations of perceived or evaluated health status are associated with ED use in cancer patients in current literature. Operational definitions of need include reasons for visits, symptoms, chief complaints, and diagnoses and vary across healthcare settings and studies. Ultimately, the underlying symptoms or diagnoses associated with ED visits are need-related determinants of ED use. Cancer patients present to the ED with a variety of complaints but tend to have symptoms or diagnoses related to pain, pulmonary, gastrointestinal, cardiac, and infectious concerns.

A systematic review of symptoms experienced by cancer patients visiting the ED identified 28 reported symptoms, including psychological (such as anxiety), gastrointestinal, neurological, respiratory, dermatological, and urological symptoms, pain, fever and infection, edema, bleeding, fatigue, and altered nutritional status [62]. The primary reasons for visits defined as chief complaints (not diagnoses) for oncologic patients tend to be related to pain, respiratory distress, fever, and gastrointestinal issues [21, 37, 45].

In recent population-based studies, diagnoses (most frequently ICD9 or ICD10) are generally used to identify "reasons" for an ED visit [10, 13, 23]. In a national sample examining cancer-related ED visits between 2006 and 2012, pneumonia was the single diagnosis associated with the most ED visits, accounting for 4.5% of cancer-related ED visits [13]. Diagnoses that each represented between 3 and 4% of visits were nonspecific chest pain, urinary tract infections, septicemia, and chronic obstructive pulmonary disease [13]. Abdominal pain, fluid and electrolyte disorders, congestive heart failure, cardiac dysrhythmia, and intestinal obstruction without hernia each represented between 2 and 3% of the diagnoses for ED visits by cancer patients [13].

Importantly, diagnoses differ by the disposition of the visit (e.g., to home vs. inpatient admission). In California, the top three diagnoses for ED visits made by individuals within 180 days of cancer diagnosis who were admitted to the hospital were septicemia (8%), cardiovascular problems (7%), and complications from surgery (5%), whereas the top three for visits that resulted in discharge home in the same sample were abdominal pain (7%), cardiovascular problems (6%), and urinary, kidney, and bladder complaints other than a urinary tract infection (5%) [23].

Regardless of visit disposition, the top ten diagnoses cumulatively account for less than 40% of all visits, underscoring the variability and complexity in precipitating factors for ED visits among cancer patients [13, 23]. Furthermore, ICD codes are limited in their ability to capture precipitating factors associated with ED use; moreover, the specific codes selected to categorize diagnoses into clinically meaningful groups vary with different algorithms used to group ICD codes to represent, for instance, surgical complications or chemotherapy-related complications.

Additional studies identify need factors associated with ED use including comorbidities, specific treatments, time to treatment, cancer stage, and survival. ED use is increased among patients having a greater number of comorbidities [43, 61, 63], a diagnosis of depression compared to no depression [64], and a radical prostatectomy compared with no treatment [56], having external beam radiation therapy with androgen deprivation therapy or external beam radiation therapy plus brachytherapy compared to radical prostatectomy [65], and having no therapy compared to radiation therapy [43, 58] and in those experiencing more severe symptoms [63], those surviving less than 1 year from diagnosis compared to surviving 1 year or more [59], and those receiving chemotherapy versus no chemotherapy [43, 61]. Cancer stage is also considered a need factor that could influence ED use. However, evidence of the impact of cancer stage on more or less ED use is unclear [11]. Additionally, among African American men with a prostate cancer diagnosis compared to white men, having a longer time to treatment was also a determinant of ED use [58].

International Perspective on Determinants of ED Use

The reasons for ED visits reported in international studies are similar to those reported in US studies. A single-site study from Brazil found the most common complaints were abdominal pain (18.4%), back pain (8.5%), dyspnea (8.5%),

weakness/fatigue (8.1%), fever (7.0%), and nausea/vomiting (4.8%) [66]. A separate single-site Australian study which focused on breast cancer patients receiving chemotherapy found the most common reasons for an ED visits were nonneutropenic fever presentations (27%), neutropenic fever (24%), pain (14%), drug reaction (10%), and infection (6%) [49]. A Canadian population-based study of ED visits for women over the age of 65 years undergoing curative surgery for nonmetastatic breast cancer identified diagnoses associated with visits to be infectious disease (19%), musculoskeletal trauma/wound (13%), other (12%), surgical site issues (12%), process of care (4%), and other noninfectious diagnoses (40%) [51]. In Taiwan, Tsai et al. found pain (27.8%), fever (11%), shortness of breath (9%), abdominal distention (4.6%), and nausea/vomiting (4.2%) to be the most frequent presenting complaints [54].

Identified predictors of ED use are more robust in some international studies, likely due to comprehensive health service use data available in nationalized health systems. In a UK study including all cancer diagnoses, Abel et al. evaluated predisposing, enabling, and need factors associated with increased ED use finding the following independent predictors: women diagnosed with cancers of the bladder, brain, colorectal, stomach, and lung; men diagnosed with oral cancer, lymphoma, and melanoma; younger age compared to older age for acute leukemia and colon, stomach, and esophageal cancers; older age compared to younger age for laryngeal, thyroid, oral, and Hodgkin lymphoma and melanoma; low SES for most cancers, especially oral, anal, laryngeal, and small intestine cancers [67].

An examination of ED visits made by cancer patients in New South Wales, Australia, found that having neurologic or lymphohematopoietic cancers (compared to the skin), having comorbidities, and living in a socioeconomically disadvantaged area increased the likelihood of an ED visit [48]. Dufton et al. identified being born outside Australia and cancer diagnoses of the head and neck, upper gastrointestinal, colorectal, lung, skin, or breast as determinants of ED use [16]. In a population-based sample of older breast cancer patients in Canada, Westley et al. reported significant determinants of ED visits, within 45 days of surgery, were localized or regional versus in situ disease, mastectomy versus lumpectomy, operation before definitive oncologic control, lower institutional volume, having more than five prescriptions, benzodiazepine use, anticoagulant use, cardiovascular disease, diabetes, past hospitalization, and lower income [51].

Preventability of ED Visits Among Cancer Patients

One of the primary purposes of examining reasons for ED visits is to determine whether or not care could have been prevented or more optimally delivered in an alternate setting.

Despite persistent attention to preventable ED use, there is no consensus definition, generally, or applied specifically to oncologic patients [11]. Two previous approaches used to identify potentially preventable or avoidable ED visits are the Agency for Healthcare Research and Quality's classification of Ambulatory Care-Sensitive Conditions (ACSC, now referred to as Prevention Quality Indicators) and the New York University's Billings algorithm [68, 69]. Both of these methods use discharge diagnoses to determine and define the reason for a preventable visit. The ACSC identifies 16 conditions that are considered potentially preventable with proper primary or ambulatory care. The Billings algorithm categorizes visits into four groups: nonemergent, emergent but primary care treatable, emergent with ED care needed but potentially avoidable if timely and effective ambulatory care had been available, and emergent when ED care was truly needed and the visit was not preventable [68]. Using such definitions, it is estimated up to 56% of all ED visits (not restricted to cancer patients) can be considered avoidable [70, 71].

Neither the ACSC nor Billings classifications are oncologic specific, and they may not adequately capture preventable use among oncologic patients. As an example, febrile neutropenia, a common chemotherapy-related side effect, is not included in either method [10, 11]. In an attempt to capture cancer-specific diagnoses, in 2016 the Centers for Medicare & Medicaid Services (CMS) established a quality metric that identifies diagnoses associated with potentially preventable ED use by cancer patients—including anemia, dehydration, diarrhea, emesis, fever, nausea, neutropenia, pain, pneumonia, or sepsis within 30 days of outpatient chemotherapy treatment [72].

Using claims data, Panattoni et al. developed a specific method to identify potentially avoidable use among oncologic patients and specified 18 diagnoses from either the CMS proposed Admissions and Emergency Department Visits for Patients Receiving Outpatient Chemotherapy Measure and/or the Symptom Tracking and Reporting PRO tool as representing potentially preventable ED visits in the year after chemotherapy, radiation, or both [10]:

- Anemia
- Appetite loss
- Constipation
- Cough
- Dehydration
- Diarrhea
- Dyspnea
- Dysuria
- Emesis
- Fatigue
- Fever
- Flushing
- Nausea

- Neuropathy
- Neutropenia
- Pain
- Pneumonia
- Sepsis

Using these potentially preventable diagnosis codes, they identified pain, fever, and dyspnea as the most prevalent potentially preventable diagnoses for cancer patients, accounting for approximately 40% of ED visits using primary diagnosis coding, and over 63% if all coding place fields were included [10].

When considering the specific diagnoses included in these methods, the concept of what is preventable remains ambiguous. Certainly, diagnoses such as sepsis, pneumonia, or dyspnea may represent clinically significant events that justify treatment and possibly admission to the hospital. Given that the diagnoses listed above may be present in over half of cancer-related ED visits but these patients might actually have serious life-threatening conditions, risk stratification methodologies are needed to truly identify patients where ED visits could be safely avoided [10, 22]. A particularly unique event to consider is patients who obtain their initial diagnosis in the ED, which is estimated to occur in 12-32% of oncologic patients [67, 73-75]. In these situations, the reason for the visit may have been directly related to symptoms that led to a diagnosis or could have been an incidental finding during the treatment of an unrelated complaint. In either circumstance, it is difficult to establish whether these visits were potentially avoidable based on diagnosis codes and even more difficult to determine if limited access to care or cancer screenings contributed to a diagnosis in the ED.

International Perspective on Preventable ED Use

Identifying potentially preventable ED visits is also challenging in international settings. Some literature, particularly from nationalized healthcare systems, focuses on a cancer diagnosis in the ED as a particular subset of preventable ED use [74, 76]. This may be a useful indicator of overall cancer care and screening in a health system, yet as described above, the ability to establish the true preventability of those visits is unclear. In a population-based Canadian study, Barbera et al. report that nearly 7% of visits were potentially avoidable and based on a definition of avoidable care that encompasses device problems, constipation, repeated prescriptions, follow-up visits, or laboratory examinations [77].

A single-site study from France specifically examined the appropriateness of referrals (based on the need for medical exam within 24 hours) to the ED for oncologic patients and the potential preventability of those visits based on presentation and treatment [78]. In this study, 33% of visits were categorized as having a high likelihood of being avoidable and 14% as having a moderate likelihood of being avoidable. Individual chart review was used to identify the driving factors for each visit (e.g., care provided at a preceding healthcare contact, an issue with a previous discharge process, patient not understanding a plan of care, etc.) to establish the likelihood of a visit being avoidable. Similarly, Oh et al. examined ED visits in a single South Korean hospital using expert review of medical records to determine if visits met the criteria they considered avoidable. They defined avoidable as a problem that could be resolved at a primary care office, out-patient clinic, or over the telephone and considered visits due to hospice referral, chemotherapy or radiotherapy, or surgery as unavoidable. Among the most common chief complaints (i.e., pain, GI symptom, dyspnea, altered mental status, other neurologic symptom, fever/chills, bleeding, edema/swelling, general weakness, urinary symptoms or constipation, and URI symptoms), the percentages of avoidable and non-avoidable visits varied within each category [15]. For example, 60% of visits for pain were classified as avoidable, while 40% were unavoidable. Likewise, 53% of visits for dyspnea were deemed avoidable, while 47% were unavoidable.

Ultimately, the ability to evaluate ED visits and the extent to which they are potentially avoidable are dependent on a country's health system and available data sources. Even in countries with robust national health systems and data collection processes, defining avoidable ED visits remains a problem, particularly for the oncologic patient population. It should be noted that approaches requiring extensive individual chart review would be difficult to apply to populationbased samples. Consensus definitions of avoidable visits and efficient, effective care systems for oncologic patients are important internationally; however, further work is needed to reach these goals.

Conclusion

One of the greatest ongoing challenges to providing highquality cancer care, for the United States and internationally, is the management of acute complications of cancer and rapidly developing treatments [79]. The need for well-coordinated care for end-of-life and palliative care is well established, but there is also a need to address acute oncologic needs [3]. Evaluating ED use by oncologic patients requires examination of both the incidence and frequency of visits by cancer patients. By understanding the volume of care provided by EDs, as well as the percentage and types of visits, health providers and systems are better able to prepare for the required treatments and partner with policy makers to establish consensus definitions of preventable and non-preventable ED use in this population. This information is essential to developing successful innovative oncologic care models which may include interventions such as telemedicine, enhanced symptom management support systems, care pathways, and oncologic-specific urgent care or emergency departments.

EDs provide a significant amount of care for oncologic patients, with 4% of all ED visits categorized as cancer related and over 40% of cancer patients having at least one ED visit within a year of diagnosis. While a few select hospitals and health systems have developed oncologic-specific EDs, urgent cares, or walk-in clinics to address unscheduled concerns, the majority of patients receiving cancer treatment must rely on general EDs for acute and unscheduled needs [3].

Determinants of ED use include a complex constellation of predisposing, enabling, and need factors, with most studied at the individual level. Less is known about broader contextual determinants of ED use in cancer patients-including provider, health system, and health policy factors-in part because of limitations in extant data to study ED use in oncologic patients. Future research is recommended to address these gaps. Based on available data, definitions of truly preventable and avoidable ED visits remain unclear. However, available data demonstrate the magnitude of potentially avoidable visits may be significant and warrants further investigation. Furthermore, examinations of reasons for visits for symptoms including pain, dyspnea, nausea, and concerns for bacterial infection may have a significant impact on enhancing cancer care, and addressing these concerns may help to reduce unnecessary ED visits [22].

References

- Institute of Medicine. Delivering high-quality cancer care: charting a new course for a system in crisis. Washington, DC: The National Academies Press; 2013. https://doi.org/10.17226/18359.
- Berwick DM, Nolan TW, Whittington J. The triple aim: care, health, and cost. Health Aff. 2008;27(3):759–69.
- Cooksley T, Rice T. Emergency oncology: development, current position and future direction in the USA and UK. Support Care Cancer. 2017;01;25(1):3–7.
- Mariotto AB, Yabroff KR, Shao Y, Feuer EJ, Brown ML. Projections of the cost of cancer care in the United States: 2010-2020. J Natl Cancer Inst. 2011;103(2):117–28.
- IMS Institute for Healthcare Informatics Global oncology trend report: a review of 2015 and outlook to 2020. Parsippany: IMS Institute for Healthcare Informatics. 2016.
- World Health Organization Global Cancer Observatory: Estimated number of incident cases from 2018 to 2040, all cancers, both sexes, all ages. 2020. https://gco.iarc.fr/. Accessed 5 May 2020.
- Prager GW, Braga S, Bystricky B, Qvortrup C, Criscitiello C, Esin E, et al. Global cancer control: responding to the growing burden, rising costs and inequalities in access. ESMO Open. 2018;3(2):e000285.
- Moore BJ (IBM Watson Health), Stocks C (AHRQ), Owens PL (AHRQ). Trends in Emergency Department Visits, 2006–2014. HCUP Statistical Brief #227. September 2017. Agency for Healthcare Research and Quality, Rockville.www.hcup-us.ahrq.gov/reports/statbriefs/sb227-Emergency-Department-Visit-Trends.pdf.
- White K, Roydhouse J, O'Riordan L, Wand T. Interventions for reducing the use of adult Emergency Department services by can-

cer patients: an Evidence Check rapid review brokered by the Sax Institute (http://www.saxinstitute.org.au) for the Cancer Institute NSW, 2013.

- Panattoni L, Fedorenko C, Greenwood-Hickman MA, Kreizenbeck K, Walker JR, Martins R, et al. Characterizing potentially preventable cancer- and chronic disease-related emergency department use in the year after treatment initiation: a regional study. J Oncol Pract. 2018;14(3):e176–e85.
- 11. Lash RS, Bell JF, Reed SC, Poghosyan H, Rodgers J, Kim KK, et al. A systematic review of emergency department use among cancer patients. Cancer Nurs. 2017;40(2):135–44.
- Brown J, Grudzen C, Kyriacou DN, Obermeyer Z, Quest T, Rivera D, et al. The emergency care of patients with cancer: setting the research agenda. Ann Emerg Med. 2016;68(6):706–11.
- Rivera DR, Gallicchio L, Brown J, Liu B, Kyriacou DN, Shelburne N. Trends in adult cancer-related emergency department utilization: an analysis of data from the nationwide emergency department sample. JAMA Oncol. 2017;3(10):e172450.
- Chen H, Johnson M, Boland E, Seymour J, Macleod U. Emergency admissions and subsequent inpatient care through an emergency oncology service at a tertiary cancer centre: service users' experiences and views. Support Care Cancer. 2019;27(2):451–60.
- Oh TK, Jo YH, Choi JW. Associated factors and costs of avoidable visits to the emergency department among cancer patients: 1-year experience in a tertiary care hospital in South Korea. Support Care Cancer. 2018;26(11):3671–9.
- Dufton PH, Drosdowsky A, Gerdtz MF, Krishnasamy M. Sociodemographic and disease related characteristics associated with unplanned emergency department visits by cancer patients: a retrospective cohort study. BMC Health Serv Res. 2019;19(1):647.
- NHS England Chemotherapy Clinical Reference Group. Clinical advice to cancer alliances on commissioning of acute oncology services, including metastatic spinal cord compression. Issue/ approval Oct 2017. https://www.cmcanceralliance.nhs.uk/application/files/5315/4279/3425/Clinical_Advice_for_the_Provision_of_ Acute_Oncology_Services_Oct_2017.pdf. Accessed 14 Aug 2020.
- Chen H, Walabyeki J, Johnson M, Boland E, Seymour J, Macleod U. An integrated understanding of the complex drivers of emergency presentations and admissions in cancer patients: qualitative modelling of secondary-care health professionals' experiences and views. PLoS One. 2019;14(5) https://doi.org/10.1371/journal. pone.0216430.
- Hargarten SW, Richards MJ, Anderson AJ, Roberts MJ. Cancer presentation in the emergency department: a failure of primary care. Am J Emerg Med. 1992;10(4):290–3.
- Jaffe D, Fleisher G, Grosflam J. Detection of cancer in the pediatric emergency department. Pediatr Emerg Care. 1985;1(1):11–5.
- Hsu J, Donnelly JP, Moore JX, Meneses K, Williams G, Wang HE. National characteristics of Emergency Department visits by patients with cancer in the United States. Am J Emerg Med. 2018;36(11):2038–1.
- 22. Caterino JM, Adler D, Durham DD, Yeung SJ, Hudson MF, et al. Analysis of diagnoses, symptoms, medications, and admissions among patients with cancer presenting to emergency departments. JAMA Netw Open. 2019;2(3):e190979.
- Lash RS, Bell JF, Bold RJ, Joseph JG, Cress RD, Wun T, et al. Emergency Department use by recently diagnosed cancer patients in California. J Community Support Oncol. 2017;15(2):95–102.
- 24. Abu-Bonsrah N, Goodwin CR, De la Garza-Ramos R, Sankey EW, Liu A, Kosztowski T, et al. Readmissions after surgical resection of metastatic tumors of the spine at a single institution. World Neurosurg. 2017;101:695–701.e1.
- 25. Bailey KL, Merchant N, Seo Y-J, Elashoff D, Benharash P, Yanagawa J. Short-Term Readmissions after open, thoracoscopic, and robotic lobectomy for lung cancer based on the nationwide readmissions database. World J Surg. 2019;43(5):1377–84.
- Goel AN, Badran KW, Mendelsohn AH, Chhetri DK, Sercarz JA, Blackwell KE, et al. Readmission after surgery for oropharyngeal

cancer: an analysis of rates, causes, and risk factors. Laryngoscope. 2019;129(4):910-8.

- O'Connell EP, Healy V, Fitzpatrick F, Higgins CA, Burke JP, McNamara DA. Predictors of readmission following proctectomy for rectal cancer. Dis Col Rectum. 2019;62(6):703–10.
- Bhagat R, Bronsert MR, Juarez-Colunga E, Weyant MJ, Mitchell JD, Glebova NO, et al. Postoperative complications drive unplanned readmissions after esophagectomy for cancer. Ann Thorac Surg. 2018;105(5):1476–82.
- Kelly KJ, Cajas L, Baumgartner JM, Lowy AM. Factors associated with 60-day readmission following cytoreduction and hyperthermic intraperitoneal chemotherapy. Ann Surg Oncol. 2018;25(1):91–7.
- Bell JF, Whitney RL, Reed SC, Poghosyan H, Lash RS, Kim KK, et al. Systematic review of hospital readmissions among patients with cancer in the United States. Oncol Nurs Forum. 2017;44(2):176–91.
- Prince RM, Powis M, Zer A, Atenafu EG, Krzyzanowska MK. Hospitalisations and emergency department visits in cancer patients receiving systemic therapy: systematic review and metaanalysis. Eur J Cancer Care. 2019;28(1):e12909.
- 32. Agency for Healthcare Research and Quality. Medical expenditure panel survey: survey background. Last revised 22 Apr 2019. https://www.meps.ahrq.gov/mepsweb/about_meps/survey_back. jsp. Accessed 1 Apr 2020.
- 33. Healthcare Cost and Utilization Project (HCUP), Agency for Healthcare Research and Quality. NEDS Overview. Last modified 27 Apr 2020. www.hcup-us.ahrq.gov/nedsoverview.jsp. Accessed 15 Aug 2020.
- 34. National Center for Health Statistics, Center for Disease Control and Prevention. National Ambulatory Medical Care Survey. Last reviewed 6 Sep 2019. https://www.cdc.gov/nchs/ahcd/about_ahcd. htm. Accessed 1 Apr 2020.
- 35. National Program of Cancer Registgires, Center for Disease Control and Prevention. How cancer registries work. Last reviewed 13 Nov 2019. https://www.cdc.gov/cancer/npcr/value/registries. htm. Accessed 1 Apr 2020.
- National Cancer Institute, National Institute of Health. Surveillance, epidemiology, and end results program. Overview of the SEER Program. https://seer.cancer.gov/about/overview.html. Accessed 1 Apr 2020.
- 37. Mayer DK, Travers D, Wyss A, Leak A, Waller A. Why do patients with cancer visit emergency departments? Results of a 2008 population study in North Carolina. J Clin Oncol. 2011;29(19):2683–8.
- Scholer AJ, Mahmoud OM, Ghosh D, Schwartzman J, Farooq M, Cabrera J, et al. Improving cancer patient emergency room utilization: a New Jersey state assessment. Cancer Epidemiol. 2017;51:15–22.
- Nasser JS, Huetteman HE, Chung TT, Chung KC. Unplanned emergency department visits within 30 days of mastectomy and breast reconstruction. Plast Reconstr Surg. 2018;142(6):1411–20.
- 40. Terzo L, Fleming M, Yechoor A, Camporeale J, Troxler M, Roth E, et al. Reducing unplanned admissions: focusing on hospital admissions and emergency department visits for patients with head and neck cancer during radiation therapy. Clin J Oncol Nurs. 2017;21(3):363–9.
- 41. Subramanian S. Impact of Medicaid copayments on patients with cancer: lessons for Medicaid expansion under health reform. Med Care. 2011;49(9):842–7.
- 42. Hassett MJ, O'Malley AJ, Pakes JR, Newhouse JP, Earle CC. Frequency and cost of chemotherapy-related serious adverse effects in a population sample of women with breast cancer. J Natl Cancer Inst. 2006;98(16):1108–17.
- 43. Weidner TK, Kidwell JT, Etzioni DA, Sangaralingham LR, Van Houten HK, Asante D, et al. Factors associated with emergency department utilization and admission in patients with colorectal cancer. J Gastrointest Surg. 2018;22(5):913–20.

- 44. National Center for Health Statistics. Health, United States, 2018 Hyattsville: Center for Disease Control and Prevention, U.S Department of Health and Human Services; 2019. https://www.cdc. gov/nchs/data/hus/hus18.pdf
- 45. Elsayem AF, Merriman KW, Gonzalez CE, Yeung S-CJ, Chaftari PS, Reyes-Gibby C, et al. Presenting symptoms in the emergency department as predictors of intensive care unit admissions and hospital mortality in a comprehensive cancer center. J Oncol Pract. 2016;12(5):e554–63.
- 46. Tanaka T, Taguri M, Fumita S, Okamoto K, Matsuo Y, Hayashi H. Retrospective study of unplanned hospital admission for metastatic cancer patients visiting the emergency department. Support Care Cancer. 2017;25(5):1409–15.
- 47. van der Meer DM, Weiland TJ, Philip J, Jelinek GA, Boughey M, Knott J, et al. Presentation patterns and outcomes of patients with cancer accessing care in emergency departments in Victoria, Australia. Support Care Cancer. 2016;24(3):1251–60.
- Bureau of Health Information. The Insights Series: Emergency department utilisation by people with cancer Sydney (NSW), Australia: BHI; 2014.
- 49. Tang M, Horsley P, Lewis CR. Emergency department presentations in early stage breast cancer patients receiving adjuvant and neoadjuvant chemotherapy. Intern Med J. 2018;48(5):583–7.
- Pittman NM, Hopman WM, Mates M. Emergency room visits and hospital admission rates after curative chemotherapy for breast cancer. J Oncol Pract. 2015;11(2):120–5.
- 51. Westley T, Syrowatka A, Henault D, Rho Y-S, Khazoom F, Chang S-L, et al. Patterns and predictors of emergency department visits among older patients after breast cancer surgery: a populationbased cohort study. J Geriatr Oncol. 2018;9(3):204–13.
- Wu V, Hall SF. Rates and causes of 30-day readmission and emergency room utilization following head and neck surgery. J Otolaryngol Head Neck Surg. 2018;47(1):36.
- Tang P-L, Cheng JS, Huang WC, Chang HS, Chen HC. Why do head and neck cancer patients visit the emergency department? Am J Emerg Med. 2015;33(8):1102–5.
- 54. Tsai SC, Liu LN, Tang ST, Chen JC, Chen ML. Cancer pain as the presenting problem in emergency departments: incidence and related factors. Support Care Cancer. 2010;18(1):57–65.
- 55. Andersen R, Newman JF. Societal and individual determinants of medical care utilization in the United States. Milbank Q. 2005;83(4) https://doi.org/10.1111/j.1468-0009.2005.00428.x.
- Jayadevappa R, Malkowicz SB, Chhatre S, Gallo J, Schwartz JS. Racial and ethnic variation in health resource use and cost for prostate cancer. BJU Int. 2010;106(6):801–8.
- 57. Hudon C, Courteau J, Krieg C, Vanasse A. Factors associated with chronic frequent emergency department utilization in a population with diabetes living in metropolitan areas: a population-based retrospective cohort study. BMC Health Serv Res. 2017;08;17(1):525.
- Jayadevappa R, Chhatre S, Johnson JC, Malkowicz SB. Variation in quality of care among older men with localized prostate cancer. Cancer. 2011;117(11):2520–9.
- Kurtz ME, Kurtz JC, Given CW, Given BA. Utilization of services among elderly cancer patients--relationship to age, symptoms, physical functioning, comorbidity, and survival status. Ethn Dis. 2005;15(2 Suppl 2):S17–22.
- Kurtz ME, Kurtz JC, Given CW, Given B. Predictors of use of health care services among elderly lung cancer patients: the first year after diagnosis. Support Care Cancer. 2006;14(3):243–50.
- 61. Goyal RK, Wheeler SB, Kohler RE, Lich KH, Lin CC, Reeder-Hayes K, et al. Health care utilization from chemotherapyrelated adverse events among low-income breast cancer patients: effect of enrollment in a medical home program. N C Med J. 2014;75(4):231–8.
- 62. Vandyk AD, Harrison MB, Macartney G, Ross-White A, Stacey D. Emergency department visits for symptoms experienced by

oncology patients: a systematic review. Support Care Cancer. 2012;20(8):1589-99.

- 63. Kurtz ME, Kurtz JC, Given CW, Given B. Effects of a symptom control intervention on utilization of health care services among cancer patients. Med Sci Monit. 2006;12(7):CR319–24.
- 64. Himelhoch S, Weller WE, Wu AW, Anderson GF, Cooper LA. Chronic medical illness, depression, and use of acute medical services among Medicare beneficiaries. Med Care. 2004;42(6):512–21.
- 65. Jayadevappa R, Lee DI, Chhatre S, Guzzo TJ, Malkowicz SB. Comparative effectiveness of treatments for high-risk prostate cancer patients. Urol Oncol. 2019;37(9):574.e11–8.
- 66. Batalini F, Gomes M, I F, et al. Cancer complaints: the profile of patients from the emergency department of a Brazilian oncology teaching hospital [version 1; peer review: 2 approved]. F1000Res. 2017;6:1919.
- 67. Abel GA, Shelton J, Johnson S, Elliss-Brookes L, Lyratzopoulos G. Cancer-specific variation in emergency presentation by sex, age and deprivation across 27 common and rarer cancers. Br J Cancer. 2015;112(Suppl 1):S129–36.
- Billings J, Parikh N, Mijanovich T. Emergency department use: the New York story. Issue brief 434. New York: The Commonwealth Fund; 2000.
- Agency for Healthcare Research and Quality. Prevention quality indicators overview https://www.qualityindicators.ahrq.gov/modules/pqi_overview.aspx. Accessed 15 May 2020.
- Weinick RM, Billings J, Thorpe JM. Ambulatory care sensitive emergency department visits: a national perspective. Acad Emerg Med. 2003;10(5):525–6.
- New England Healthcare Institute. A matter of urgency: reducing Emergency Department overuse. NEHI Research Brief March 2010. Boston MA. https://www.nehi.net/writable/publication_files/ file/nehi_ed_overuse_issue_brief_032610finaledits.pdf. Accessed 1 Apr 2020.
- 72. Center for Medicare & Medicaid Services. CMS measures inventory tool. Admissions and Emergency Department (ED) Visits for Patients Receiving Outpatient Chemotherapy. 2020. https://cmit. cms.gov/CMIT_public/ViewMeasure?MeasureId=2929. Accessed 17 Aug 2020.
- Livingood WC, Smotherman C, Lukens-Bull K, Aldridge P, Kraemer DF, Wood DL, et al. An elephant in the emergency department: symptom of disparities in cancer care. Popul Health Manag. 2016;19(2):95–101.
- 74. Rogers MJ, Matheson LM, Garrard B, Mukaro V, Riches S, Sheridan M, et al. Cancer diagnosed in the emergency department of a regional health service. Aust J Rural Health. 2016;24(6):409–14.
- 75. Murchie P, Smith SM, Yule MS, Adam R, Turner ME, Lee AJ, et al. Does emergency presentation of cancer represent poor performance in primary care? Insights from a novel analysis of linked primary and secondary care data. Br J Cancer. 2017;116(9):1148–58.
- 76. Abel GA, Mendonca SC, McPhail S, Zhou Y, Elliss-Brookes L, Lyratzopoulos G. Emergency diagnosis of cancer and previous general practice consultations: insights from linked patient survey data. Br J Gen Pract. 2017;67(659):e377–e87.
- 77. Barbera L, Atzema C, Sutradhar R, Seow H, Howell D, Husain A, et al. Do patient-reported symptoms predict emergency department visits in cancer patients? A population-based analysis. Ann Emerg Med. 2013;61(4):427–37.e5.
- Duflos C, Antoun S, Loirat P, DiPalma M, Minvielle E. Identification of appropriate and potentially avoidable emergency department referrals in a tertiary cancer care center. Support Care Cancer. 2017;25(8):2377–85.
- 79. Young A, Marshall E, Krzyzanowska M, Robinson B, Brown S, Collinson F, et al. Responding to acute care needs of patients with cancer: recent trends across continents. Oncologist. 2016;21(3):301–7.

Models of Care

Michael G. Purcell

Introduction/Background

Both the emergency medicine and oncologic communities recognize that cancer patients require specialized emergency care and are better served by professionals who are knowledgeable about their unique needs. Within emergency medicine, this is highlighted by the relatively recent formation of the Society for Academic Emergency Medicine Oncologic Emergencies Interest Group and the Comprehensive Oncologic Emergencies Research Network (CONCERN). Patients often relate stories of being told in their local emergency department (ED) to go to their cancer center for further treatment after emergent conditions have been excluded. Conversely, oncologists rarely have access to EDs with specific oncologic expertise. Patients express concern that emergency physicians in the community are not completely comfortable caring for complex oncologic patients and lack adequate knowledge regarding the management of their disease processes and treatments. As a result of their patients' prior experiences in these less specialized settings, oncologists are often hesitant to recommend such EDs with limited oncologic expertise to their patients. Many oncologists who work in large centers are requesting urgent and emergent after-hours services by personnel who are trained in handling oncologic emergencies. With overcrowding and prolonged waits for treatment that characterize many of our nation's EDs, those with cancer and complex care needs, including immunocompromised, intractable pain, and end-of-life care needs, may best be served in regionalized EDs specializing in oncologic care.

The numbers of cancer patients and survivors among the general population are increasing. The life expectancy of

cancer patients has increased significantly in the last six decades. In 2017, there were an estimated 15,760,939 persons living with cancer in the United States alone. In 2020, an estimated 1.8 million additional Americans will be diagnosed with cancer. The 5-year survival rate was estimated to be 67.4% between 2010 and 2016 [1]. With the advent of new therapies and treatment modalities, this survival rate could conceivably increase.

Comparative survival data from the MD Anderson Cancer Registry (The University of Texas MD Anderson Cancer Center, Houston), established in 1944, demonstrate a marked improvement in survival rates for most malignancies. Examples include breast cancer, the 10-year overall survival rate having increased from 25% in 1944 to 76.5% in 1995 for patients treated at MD Anderson. For prostate cancer, the most common malignancy in men, the 10-year survival rate increased from 8.5% in 1944 to 82.5% in 1995. Acute myeloid leukemia was simply fatal in 1944, with a median survival from diagnosis of 8 weeks and a 99% mortality rate at 12 months, but by 2004, the long-term survival rate had increased to over 25%. Remission rates in acute myeloid leukemia patients under age 60 years have reached 65% [2]. Thus, there are many cancer survivors seeking medical care in primary care offices and EDs around the country.

To further highlight the need for such specialized care, there will be an estimated 26 million persons in the United States who either have active cancer or who have been previously diagnosed with cancer in 2040 [3]. Seventy-three percent of this population will be over the age of 65. This ever-growing and aging population will continue to seek emergency care both during and following the diagnosis and treatment of their cancer. Currently, about 4% of all ED visits are made by patients with a cancer diagnosis [4]. This population has a high rate of admission, 59%, when compared to the 16% admission rate for those without cancer [3]. This stems from infection, therapy side effects, manifestations of the malignancy itself, as well as a plethora of hematologic and metabolic derangements. With increasing prevalence, 5-year survival rate, and age of the patient, there

M. G. Purcell (🖂)

Department of Emergency Medicine, Ohio State University Wexner Medical Center, Columbus, OH, USA e-mail: Michael.Purcell@osumc.edu

K. H. Todd et al. (eds.), Oncologic Emergency Medicine, https://doi.org/10.1007/978-3-030-67123-5_2

will be an increasing need to have providers more familiar with cancer and its therapy. This may be best accomplished by personnel with additional experience and training in treating and managing this population.

Several other factors have increased the population of oncologic patients and survivors seeking acute care. In the last few years, more oncologic patients have been receiving treatment as outpatients. Leukemia and stem cell transplant patients spend less time in the hospital and often receive the majority of their chemotherapy in outpatient treatment centers. These patients are no longer universally admitted to the hospital for neutropenia if there is no evidence of infection. Instead, they make frequent trips to the hospital for treatment and laboratory evaluations. Often, patients arrange temporary housing in the area of the oncologic treatment center. This practice has also increased the need for unscheduled acute care. Furthermore, cancer patients and survivors have a combination of medical problems that may or may not be related to their cancer history and a wide range of potential residual medical issues related to their prior disease and/or treatments. Meanwhile, oncologic care is becoming increasingly specialized. Oncologic practice is focusing on emerging treatments and targeted therapies. As more treatment options become available, more expertise is needed in each oncologic subspecialty. With increasing treatment options, there are more potential side effects and treatments available for the supportive care of these patients.

Cancer patients not only suffer from complex medical problems related to their disease and therapy but also are particularly vulnerable emotionally. Patients suffering from a life-threatening illness often have stronger bonds with their medical providers that may be associated with higher expectations for care and an increased sensitivity to their care providers' words or actions; conversations can take on a greater meaning and become more emotionally charged than under normal circumstances [5].

Caring for patients with advanced cancer is stressful for clinicians, and discussing bad news often evokes strong emotional feelings. Not all physicians are formally trained for these difficult communication tasks. End-of-life talks are time-consuming and stressful in any environment, but this is compounded in the ED, where the cancer patient's needs must compete with the treatment demands of other patients. Furthermore, delirium may be far more common in this population than previously thought, which may prevent meaningful discussions in the ED [6]. Unfortunately, evaluation in the ED often reveals progression of the underlying malignancy and may raise the topic of transition to palliative care. Most emergency physicians feel ill-equipped to have this discussion due to the brief nature of their relationship with the patient and lack of depth of understanding of the patient's disease and its progression and possible therapeutic options. At the same time, the patient, faced with new knowledge about disease progression manifested by the symptoms that prompted the emergency visit, may have multiple questions and a high level of anxiety. At this time, the patient is at high risk for feelings of abandonment, especially if the emergency physician is unable to answer questions or provide adequate reassurance that the patient's primary oncologist will be available to them in a timely fashion [5]. Nursing staff may also be unprepared to care for patients who are actively dying, and they may lack the skills to manage end-of-life symptoms.

Despite these needs, there are very few acute care facilities dedicated entirely to the care of cancer patients. MD Anderson and Memorial Sloan Kettering Cancer Center (New York) have such centers. The Ohio State University Wexner Medical Center has an integrated Oncologic Pod within its main ED. Other institutions with a large percentage of oncologic patients are developing resources to provide the specialized care these patients need and to mitigate the difficulties these patients can present to a busy ED. Some institutions are opening fully integrated cancer units within their EDs. They are examining ways to quickly recognize acutely ill oncologic patients so that high-risk patients are treated expeditiously while maintaining an appropriate triage system so that other patients do not perceive oncologic patients as receiving preferential treatment [7].

In this chapter, we describe several models for providing care for oncologic patients in the emergency setting. The models range from EDs at large, dedicated cancer centers (MD Anderson and Memorial Sloan Kettering) to a fully integrated Oncologic Pod that resides within a more traditional ED (The Ohio State University Wexner Medical Center, Columbus, Ohio). These models illustrate some of the pivotal issues of institutions embarking on this endeavor.

Table 2.1 Common issues essential to all oncologic ED models

Recognition and expeditious treatment of oncologic emergencies such as neutropenic fever, spinal cord compression, tumor lysis syndrome, and pulmonary embolism

Appropriate management of pain for patients who are not opioid naïve

Management of frequently needed procedures such as thoracentesis and paracentesis

Early recognition and proper management of patients who have do not resuscitate (DNR) orders or are near the end of life

Knowledgeable management of complications of cancer treatment (chemotherapy, immunotherapy, radiation, CAR-T, and other novel therapies)

Proper communication regarding disease progression with the patient and oncologist

Adequate support from end-of-life services such as palliative care and hospice

Consistent and reliable method of communication with the patients' oncologists

Support for patients who are new to the institution and attracted by the cancer ED designation
Table 2.1 lists common issues that are considered essential to all models of care. Different models for providing emergency care to cancer patients are derived from the variable needs and characteristics of each practice, such as the prevalence of cancer types, the physical and administrative organization of the local oncologic services, and the resources available.

The University of Texas MD Anderson Cancer Center

The Emergency Center at MD Anderson Cancer Center is dedicated exclusively to the care of cancer patients. It is located in the main hospital building and is designated a level III ED by the The Joint Commission and the Centers for Medicare & Medicaid Services. Ninety-eight percent of patients treated in MD Anderson's ED have cancer or a cancer history. The ED has a large role in the inpatient services provided at MD Anderson. Thirty-nine percent of hospital admissions come through the ED [8].

However, MD Anderson did not open its doors with an ED in place. The ED developed gradually as a response to the needs for acute care for the large number of outpatients being treated at MD Anderson. Initially, urgent and emergent services were provided in an open ward. No doctors were assigned to the area, and when a patient requiring emergent care arrived, the patient's physician was notified and sent to the ward to evaluate the patient. This situation was not optimal for acutely ill patients or for patients scheduled in the clinic, and the lack of individual patient rooms made it difficult to maintain patients' privacy and confidentiality [9]. The system was also disruptive for oncologists, who already had full clinical schedules. Eventually, full-time physician coverage was established, initially provided by the Department of General Internal Medicine. In 1986, the ED was formally opened. Initially, it had 23 private rooms and provided care to approximately 14,000 patients per year. In 2007, the emergency center moved to its current expanded location. In 2011, MD Anderson established an academic Department of Emergency Medicine, the first such department dedicated entirely to oncologic emergency medical care, education, and research. The MD Anderson ED currently has 45 private rooms, a six-chair unit, and a two-chair triage bay. The ED is equipped with two resuscitation rooms in which critical care is provided to patients with high acuity that arrive from the clinics, walk-in, or arrive by ambulance. The ED now sees over 26,000 patient visits annually. All of the patients have cancer or are cancer survivors, except for an occasional family member of a patient or an employee.

The ED is staffed with full-time faculty members, the majority of whom are board certified in internal medicine or emergency medicine. Some faculty members are board certified in surgery, pediatrics, or infectious disease or palliative medicine. The physicians are faculty at the University of Texas and have similar academic obligations for research, administration, and teaching as other MD Anderson faculty members. The Department of Emergency Medicine recently initiated an oncologic emergency medicine fellowship, now in its eighth year. Mid-level providers are utilized in the ED, but provide a relatively small portion of the care delivered.

The department's 19 faculty members provide round-theclock coverage. Coverage ranges from two to six physicians with an additional mid-level provider at the busiest times. The ED employs approximately 75 registered nurses with a nurse-to-patient ratio of approximately 1 to 3.

Care and treatment decisions are made by the emergency medicine faculty. However, oncologists do provide a call schedule, and there is frequent communication on an asneeded basis between the emergency physicians and primary oncologists. Oncologists do not routinely round in the ED unless they have admitted patients boarding there. The electronic medical record provides full access to the patient's medical record. Oncologists can notify the ED staff of a patient's pending arrival with the addition of important clinical information by entering a note in the medical record. After patients are seen, a note is generated by the emergency physician notifying the primary oncologist that the patient was seen. If consultation is warranted, the oncologist is contacted by phone.

The average ED length of stay is just over 6 h for a nonadmitted patient and over 9 h for an admitted patient. The ED admits 51% of the patients presenting for treatment. Approximately 30% of unique patients have hematologic tumors (leukemia or lymphoma) or have received stem cell transplantation, comprising 50.3% of all patient visits; the remainder have solid tumors [7].

Of all the patients visiting the ED in 2010, hematologic patients averaged 2.2 visits per patient, and solid tumor patients averaged 1.8. Of these patients, 12% had four visits or more, with a range of 1–31 visits per patient. Most patients were receiving multiple medications and presented with several complaints. The complexity of their illness and frequent requirements for intravenous fluids, antibiotics, electrolytes, and blood products results in a prolonged length of stay compared to other EDs. The high level of acuity is reflected in the 10.9% mortality rate associated with admission of these patients [7]. The mortality rate is higher for patients with hematologic tumors (13.6%) than for patients with solid tumors (9.8%).

Patients present to the oncologic ED with a multitude of different complaints. At MD Anderson, the most common chief complaint is fever, present in 23% of patients. This is closely followed by abdominal pain, generalized pain, shortness of breath, nausea and vomiting, weakness and fatigue, back pain, chest pain, bleeding, cough, and diarrhea.

Memorial Sloan Kettering Cancer Center

Memorial Sloan Kettering has an Urgent Care Center (UCC), dedicated solely to the care of oncologic patients. The number of patient encounters per year in the UCC has steadily increased from 14,800 in 2000 to 21,800 in 2013. Although the UCC receives Memorial patients who arrive from the community via ambulance, general 911 calls from the community are not brought to Memorial. The physical size of the unit has grown over time. Originally an eight-bed unit with an adjunct clinic space, the UCC now consists of 19 telemetry beds and 4 transfusion chairs. Turnover of these beds occurs more than four times per day.

The driving forces behind this growth are an increase in the number of patients receiving treatment at Memorial Sloan Kettering and the continued transition of oncologic care away from the inpatient setting. As cancer treatment paradigms change, the UCC is key to the institution's ability to provide acute evaluation and management to an increasingly large and complicated outpatient population. The recent addition of a freestanding same-day surgical center and the continued expansion of the outpatient bone marrow transplantation program are examples of the trend toward outpatient treatment of cancer patients.

The clinical staff consists of 13 full-time board-certified internal medicine physicians, some of whom have completed subspecialty training in palliative care, anesthesia/critical care, and infectious disease. UCC physicians are considered academic faculty who are responsible for teaching medical students and residents from Weill Cornell Medical College, as well as participating in clinical research.

Patients treated in the UCC reflect the spectrum of disease seen at Memorial. Most patients have solid tumors (72%) and are evaluated for acute complications of their disease and treatment. The most common chief complaints include dyspnea (17%), fever (14%), pain (11%), nausea (10%), and fluid/electrolyte disturbances (9%). The average length of stay in the UCC is 4 h, and slightly more than half of the patients seen in the UCC will require admission to the hospital. Occasionally, patients with advanced disease who have been treated at other institutions or individuals with a suspected but unconfirmed cancer diagnosis seek to transfer their care by visiting the UCC. Emergent problems are acutely managed; however, referral for expedited outpatient evaluation is the preferred pathway, as the UCC is not intended to be the first point of contact for a new patient.

The UCC has attempted to integrate successful models of care from emergency medicine as volume and throughput have increased. A modified Emergency Severity Index (ESI) tool is used for triage. Patients are assigned a score of 1–5 based on the need for a lifesaving intervention, the presence of a high-risk situation, the number of resources a patient will require, and predefined vital signs. Specific triaging emphases that reflect the unit's focus on oncologic include

with the rapid identification any of the following conditions: recent bone marrow transplantation, febrile neutropenia, and potential spinal cord compression. During peak hours, a UCC physician assists the triage nurse, a model that has been associated with faster throughput and improved patient outcomes in non-cancer EDs [10].

As many patients are referred internally by treating oncologists and surgeons, an electronic "UCC Notification Order" allows these individuals to communicate the most likely diagnosis, the need for admission, and which tests and consultants will expedite care.

Oncologic patients have an inherent risk for developing sepsis. An institutionally derived algorithm is used to screen all electronically documented vital signs for sepsis. When potentially significant abnormalities are identified, an alert is triggered, prompting a clinician to assess the patient for the possibility of sepsis. This process is time sensitive and requires the clinician to either document a reason for exclusion (dehydration, arrhythmia, end-of-life/palliative care, etc.) or acknowledge the alert and initiate the sepsis management protocol within 30 min.

Patients who arrive critically ill and in need of an immediate intervention such as endotracheal intubation, cardiopulmonary resuscitation, or initiation of vasopressor support are frequent challenges in cancer EDs. At Memorial Sloan Kettering, the primary oncologist has often already established and documented the goals of care in the electronic medical record. If the patient has previously consented to a do not resuscitate (DNR) order, this information is displayed in the header at the top of the screen, next to the patient's name and medical record number. This order must be confirmed and renewed with each hospitalization, as per New York State law. For critically ill patients without previously established advanced directives, the UCC clinician will rapidly determine the goals of care with the patient, healthcare proxy, and primary physician at MSK. For individuals who decline life-sustaining interventions, the UCC clinician will enter a DNR order and initiate palliative care. Preexisting order sets for narcotic analgesia and a palliative care consultant facilitate care. A medical ethics consultation service is available 24 h a day for encounters in which the goals of care are difficult to establish.

A Fast-Track Pathway is used for patients with a low Emergency Severity Index (ESI) score. One of the most common diagnoses in this group is a new, suspected, or incidentally identified thromboembolic disease. If anticoagulation is indicated, the patient is often discharged on rivaroxaban with close follow-up in the Anticoagulation Management Clinic.

In July 2013, the UCC opened an observation unit, intended for patients who were unsuitable for discharge but had an expected duration of care lasting less than 24 h. Although the observation unit is physically located in the hospital, this nine-bed unit is considered an outpatient service and is staffed by UCC physicians and mid-level provid-

ers. During the first 6 months of the program, roughly 10% of UCC visits (n = 1013) resulted in patient placement in the observation unit. The proportion of admissions to the hospital from the UCC with a length of stay less than 24 h dropped significantly after observation unit implementation (2.4-1.1%). The most common reasons for observation unit placement are fluid and electrolyte disorders (14%), pain control (14%), dyspnea (13%), and fever (9%). Interventions for patients in the observation unit include placement or revision of drainage catheters (pleural, biliary, genitourinary tract, abscess); endoscopy and transfusion in patients with hemodynamically stable gastrointestinal bleeding; correction of uncomplicated electrolyte derangements; administration of intravenous (IV), antiemetics, IV antibiotics (for treatment of cellulitis, pneumonia, and uncomplicated febrile neutropenia), or IV analgesia; and the management of severe constipation. Approximately one-third of patients placed in the observation unit require admission to the hospital for ongoing care. Extending the observation period to 48 h may decrease this number.

Approximately 15 patients a week are seen in the UCC for elective palliative paracentesis, performed by the UCC clinical staff. Drainage of symptomatic pleural effusions is performed in the observation unit by pulmonary medicine. Patients with low-risk febrile neutropenia are either discharged or placed in the observation unit for 24 h.

When possible, management decisions are made with input from a patient's primary oncologist or surgeon, who is notified automatically by e-mail during check-in and discharge. While these individuals may be off-site, they are able to review all relevant clinical data, including lab findings, chart notes, and radiology and telemetry results. An electronic status board, visible on all computer terminals within the institution and on overhead monitors in the UCC, facilitates a quick grasp of key metrics related to an individual patient and overall throughput at any given time. This tool facilitates communication about arrival and waiting times, identification of treating or covering UCC staff, pending diagnostic tests and consultants, disposition (admitted/discharged/observed), and bed status.

The Ohio State University Comprehensive Cancer Center: Arthur G. James Cancer Hospital and Richard J. Solove Research Institute

In April 2015, the Ohio State University Wexner Medical Center (OSUWMC) opened a new, specialized pod in the ED to care for its cancer population. The ED, which houses the newly named Oncologic Pod, currently cares for all patients that arrive to OSUWMC seeking emergency care: approximately 82,000 patients per year. The Oncologic Pod curTable 2.2 Challenges to an integrated Oncologic Pod

Early identification of the hematology and oncologic patient
Equitable triage and placement for all ED patients
Identification of febrile neutropenia and other subtle, life-threatening
oncologic emergencies
Waiting areas for the immunocompromised patient
Available bed space
High ESI level in the cancer population
High admission rate in the cancer population

rently evaluates, manages, and treats approximately 11,000 oncologic and hematology patients per year, which reflects over 13% of emergency visits to the OSUWMC ED. With opening of the Oncologic Pod, the ED dealt with many challenges as illustrated in Table 2.2, beginning with patient identification. The hematology and oncologic patients are identified immediately upon arrival during the triage process. Patients who arrive to the ED are asked two screening questions: "Have you seen a cancer doctor or doctor at the James in the last 12 months?" and "Are you currently undergoing active treatment for cancer?". An affirmative answer to either question allows for preferential placement into the Oncologic Pod, which is fully integrated within the ED.

One of the many challenges that the ED initially faced was the development of triage criteria to effectively triage cancer patients to the Oncologic Pod while maintaining equity among all patients that presented for evaluation. The Oncologic Pod originally opened with 10 beds and 5 additional chairs for a total of 15 treatment spaces that were allocated to the care of cancer patients. Ten of the rooms were private, four had private bathrooms, and two had negative airflow. The other five spaces were treatment bays with lounge chairs for infusions. With rising acuity and increasing number of cancer patients arriving to the ED, these treatment bays were renovated to include telemetry and actual ED beds, instead of the initially planned chairs. Additionally, the ability to flex up to 19 treatment spaces was created through the addition of four hallway beds. Patients may be placed in a hallway bed to facilitate early treatment while awaiting placement in a room or treatment bay. On days when a larger number of oncologic and hematology patient visit than the 15-bed/19-treatment space pod can accommodate, additional patients will be evaluated in the remainder of the ED when space is available. If a high acuity patient arrives to the ED and the Oncologic Pod is full, then that patient may be placed in a bed outside of the Oncologic Pod to facilitate prompt treatment of the emergent medical condition. Similarly, when there are fewer Oncologic Pod patients, non-cancer patients will be evaluated as needed in the Oncologic Pod. This will ensure equal access to emergency care for all patients, regardless of their disease state.

After the initial triage process, patients are either placed in an available treatment space or escorted to the waiting room. The waiting room represented an additional challenge. With the steadily rising number of ED visits by cancer patients, concerns arose in placing what could certainly be an immunocompromised population in the main ED waiting room. The main ED population often sought care for viral or bacterial illnesses. Such illnesses could prove life threatening for the immunocompromised cancer patient. Out of concerns for patient safety, an additional waiting area for cancer patients was created that allowed for better isolation and distancing. The cancer population viewed this as a significant improvement in their ED encounters. Hand sanitizer and facial masks are readily available for patient use in this area.

After patients are placed into a treatment space, they are cared for by a multidisciplinary team in the Oncologic Pod. This team is composed of physicians, advanced practice providers, nurses, patient care associates, patient experience representatives, social workers, case management, and dedicated emergency medicine trained pharmacists. All physicians that care for these patients are either board certified in emergency medicine or board eligible for the American Board of Emergency Medicine certifying board exam. The physician group provides 24/7 oversight of the Oncologic Pod. There are 16 hours (two, 8-hour shifts) of dedicated physician coverage in the Oncologic Pod. This runs from 9 a.m. until 1 a.m. During this time period, the dedicated physician is responsible only for care in the Oncologic Pod. Additional physician staffing throughout the department enables this physician to dedicate all of their time on shift to the cancer population. From 1 a.m. until 9 a.m., an emergency physician provides oversight in the Oncologic Pod as well as an adjacent ED area. The decision for this staffing model was based on ED arrival times of the cancer population at the OSUWMC ED, which consistently demonstrated fewer arrivals in the 1 a.m. to 9 a.m. time period on all days of the week.

To assist the physician in caring for patients, a group of advanced practice providers (APPs) staff the Oncologic Pod. This group, a mixture of both nurse practitioners and physician assistants, is dedicated solely to the care of patients in the Oncologic Pod. The APP staff provide 48 hours of coverage daily in the Oncologic Pod. This is broken down into four, 12-hour shifts with overlapping coverage. During the onboarding process, the APP staff are cross-trained in the ED and the cancer center. This includes time in the Oncologic Pod as well as rotating with the hematology, oncologic, and neuro-oncologic services. Depending on provider preference, they may also spend time with radiation oncologic or one of the many surgical services for the hospital. The offservice onboarding process prepares the APP staff for the variety of cancers and treatments that they may encounter in their role as providers. Their time onboarding in the ED further prepares them for the variety of presentations that they may encounter in their role. The APP team evaluates the vast

majority of patients in the Oncologic Pod. If there is an influx of patients, then resident physicians, who staff the adjacent pod, are readily available to assist in evaluating the cancer population. Additionally, they have 24/7 access to a boardcertified emergency physician. Through the combination of onboarding, monthly meetings, CME, and personal education, the APP staff is more than adequately prepared to deal with any oncologic emergency that comes through the door.

The work of the providers would be naught without additional staff. The Oncologic Pod has a dedicated nursing staff. The vast majority have had training either in the care of the oncologic patient or in an ED. The nursing staff is acutely aware of the presence of ports and use of other intravascular access devices. They are attuned to the needs of this particular population including aggressive symptom control and need for expeditious evaluation of a fever. The nursing staff is aided by patient care associates, up to three at a time, who help with additional tasks in the area. To help complement the immediate patient care side, 24-hour social work is available for the patients. Every cancer patient who is roomed in the ED is evaluated by a social worker to discuss living situations, safety, and advanced directives. The social work team is readily available to assist the population around the clock. Case managers also help with coordination of care. They are available to connect the differing care teams as well as to establish appointments for patients being discharged from the Oncologic Pod. Finally, the Oncologic Pod has a dedicated pharmacist, trained in emergency pharmacology. They help with a variety of issues, including antibiotic selection, antibiotic dosing, symptom control, and any other pharmacotherapeutic questions the treatment team may have. They are a valuable resource as cancer treatments continue to evolve.

One important scenario to emphasize is the patient with neutropenic fever. These patients are often difficult, but critical, to recognize. Current guidelines recommend that these patients receive antibiotics within 1 hour of triage and be monitored for 4 hours following antibiotic administration [11]. Many of these patients may appear well and traditionally have had to wait with other patients for further evaluation. Unfortunately, a prolonged time to antibiotics can result in deterioration and development of sepsis. To improve the management of these patients, the Oncologic Pod has the criterion that any patient with a fever who has received chemotherapy or radiation in the prior 2 weeks will be evaluated under the ED Sepsis Alert process. This process brings together a multidisciplinary team (physicians, nursing, radiology, pharmacy, etc.) to expedite initiation of IV antibiotics and diagnostic work-up for this high-risk population of patients.

As treatments advance, there is the ongoing need for increasing flexibility in triage. With the FDA approval of CAR-T therapy, new challenges arose. For this reason, any patient on CAR-T therapy in the James Cancer Center is provided with a card to present upon arrival to ANY ED. This details their therapy and possible side effects including complement release syndrome and neurologic side effects. The card also contains a number for outside hospitals to call for guidance on treatment. At the OSUWMC ED triage, a third screening question was recently added to help identify this population upon ED arrival. Other oncologic emergencies may be harder to identify. However, with the Oncologic Pod screening pod questions at triage, patients are immediately flagged as cancer patients upon arrival. This gives the Oncologic Pod physician and APPs, as well as the provider in triage, the opportunity to review the patient chart, chief complaint, and triage note. These providers are able to work with the triage nurses and charge nurses to expedite care of the cancer population including ordering CT scans, labs, or symptom control in triage. They can also increase or decrease the ESI or recommend that a certain patient get the next available treatment space if there is a concern for other subtle oncologic emergencies.

Other challenging scenarios that the Oncologic Pod has encountered include the arrival of patients without a clinical cancer diagnosis. An inpatient service was designed that handles the care for patients without a definite cancer diagnosis but identified as being at high risk for malignancy (i.e., new, large lung mass or abdominal mass). This facilitates the care of patients with a presumed diagnosis of cancer who may be attracted to the cancer ED, either based on outpatient imaging or as transfers from outside hospitals. This allows for patients who are not already receiving their cancer care at the James Cancer Center to be seen in the Oncologic Pod to facilitate transition of their care to the cancer center. Additionally, if patients do not require hospital admission for this work-up, a James Diagnostic Clinic can facilitate an expedited work-up for outpatients. The Oncologic Pod serves as the first point of contact of the James Cancer Center for one to two patients per day, so this is not an unusual scenario. While there is month-to-month fluctuation, this number has generally increased since the genesis of the Oncologic Pod in 2015.

As patient volumes rose over time, it became apparent that there was not only a growing need for emergent cancer care but also the need for acute, unscheduled visits. The acute, unscheduled visit encompassed patients who might need to see a provider, though do not necessarily need an ED encounter. This could range from fever in patients not on cytotoxic chemotherapies, anemia, thrombocytopenia, electrolyte abnormalities on routine labs that need follow-up, ED follow-up, or even clinic overflow when patients are not able to see their primary team. This necessitated the development of an additional eight-bed treatment space, the Immediate Care Clinic (ICC). The ICC is a 24/7 treatment space that opened in 2018. Patients established with the James Cancer Center, or in the process of establishing care, can be referred to the ICC by their provider, the ED, or the nursing triage line. Certain exclusion criteria were created to prevent those with true emergencies from arriving at the ICC. Such patients are redirected to the ED for emergent care. The ICC is staffed by the same APP group as the Oncologic Pod. It is overseen by the Oncologic Pod emergency physician. The ICC admission rate is consistently below 30%. It serves as an intermediary between the ED and outpatient worlds. This allows for lower acuity patients to be seen promptly when ED volumes are high and prevent unnecessary ED visits, increased charges to the patient, and increased overall healthcare costs.

It is one thing to establish an Oncologic Pod for cancer patients, and yet another to ensure that quality standards are being met. In order to assure that patient care is performed at the highest professional standards, there are regular evaluations of specific ED metrics. This occurs both at the level of the Department of Emergency Medicine and at the hospital level. A monthly scorecard containing information such as patient arrivals, door to provider, length of stay, admission rate, and new patient contacts is disseminated to the administration of both the hospital and ED. It is regularly reviewed by the Department of Emergency Medicine administration in conjunction with nursing leadership. It is also reviewed semiannually at the hospital level in forums such as the Patient Quality, Safety, and Reliability Committee and the Medical Staff Advisory Committee and with the Chief Medical Officer of the James Cancer Center. To ensure that patient voices are heard, patient experience representatives are available to speak with patients. Additionally, all patients who arrive are provided with a direct phone number to the patient experience representative line in the event they wish to express gratitude or concern about their ED stay. These accolades and concerns are reviewed and addressed on a weekly basis in a multidisciplinary meeting including patient experience, physician leadership, and nursing leadership. These measures ensure that high-quality care is being delivered to the patients and that all needs and expectations are met.

Current and Future Considerations for the Cancer Emergency Department

Increasing specialization has resulted in a fragmentation of medical care and cancer care is no exception. Many oncologic patients are treated by several physicians who are all specialists in cancer therapy. One patient may have one or more surgeons, a medical oncologist, a radiation oncologist, and a palliative care physician. This does not include other specialists for chronic issues or problems that develop during treatment, including cardiologists, nephrologists, endocrinologists, and pulmonologists. Patients are often confused as to which doctor is "in charge" and whom to ask which question. The role of the emergency physician in a comprehensive cancer center has some similarities to that of a primary care physician. The ED physician often explains the roles of the different providers and facilitates communication between the various specialties involved in the patients' care. Another important role is that of a safety net, by providing care to the patients when they cannot wait for an office visit or when the office visit results in the discovery of a problem that is beyond the scope of the oncologist or specialist. In these roles, the ED supports both oncologists and patients. Physicians specializing in oncologic emergencies use unique skills and knowledge of potentially dangerous complications of different treatment modalities and the best supportive therapies as well as understanding of the disease process of multiple different malignancies and their associated emergencies. Also valuable are expertise in pain management, procedures commonly needed in cancer patients, and skillful management of palliative and end-of-life care. This skill set, which currently can only be obtained through experience, helps doctors who specialize in the acute care of cancer patients make decisions regarding the aggressive or supportive nature of care provided in the cancer ED.

Several themes are prevalent in the acute care of cancer patients. One of the concerns expressed by physicians seeking to provide acute care to oncologic patients is access to the complete medical record and the expertise of the oncologist. The ED physicians must have a significant understanding of the treatment paths and modalities of the patients they are seeing. In order to make appropriate decisions, communication must be available with the oncologist and other supportive services. With more knowledge and experience, the emergency physician can be more effective in support of the patients and the oncologists and be more confident in their independent decision-making. A method of documentation and a process of communication that make the primary oncologist aware of all visits to the ED are optimal. At MD Anderson, an online medical record documents the visit and outcome, and is accompanied by an e-mail notifying the oncologist of the emergency visit, closing the communication loop. Memorial Sloan Kettering has gone one step further by posting the ED tracking board throughout the institution. At the The Ohio State University Wexner Medical Center, oncologic teams are notified of emergent visits by e-mail. They are also available to discuss patient care in real time through a variety of modalities.

Another common concern is that caring for this group of patients is very labor intensive. These patients are often very ill; many of them are not independently ambulatory. Most of the patients are on multiple medications and have numerous comorbidities and several complaints. Due to the complexity of their illness, their stay in the ED is longer than that of other populations. Many of the patients require electrolytes or blood replacement as an incidental finding or the reason for the visit. These processes add to the time in the ED and the nursing workload. The ubiquitous admission rate of over 50% and the high mortality rate of patients admitted through the ED are further testimony to the high acuity level of the patients [3].

An ED that treats only cancer patients does not have to devise a triage method to identify the cancer patients from the non-cancer patients, and recognition of neutropenic fever, sepsis, and infection with underlying immunocompromised is routine. Other problems, such as managing intractable pain and mixing and adjusting large doses of opioids, are a frequent occurrence. However, these are issues that EDs—who want to support a large cancer population but cannot be dedicated solely to that population-contend with. A frequent issue more unique to a cancer ED is the arrival of patients with a recent diagnosis of suspected or confirmed malignancy. One of the challenges of working in a cancer ED is handling a group of patients with varying degrees of illness, varying knowledge about their condition, and different stages of diagnosis who have recently received difficult news and are emotionally charged. In all of the functioning cancer EDs interviewed, avoiding having the cancer ED serve as the intake portal for the cancer institute has been a common theme. Another frequent challenge is patients with late-stage cancer with no prior relationship to the parent institution. Many of these patients have received treatment at other centers and when told that no further treatment options exist, go to the cancer ED hoping for a salvation therapy. These patients are often too sick to be discharged and, without the evaluation of an oncologist in the emergency center, will ultimately be admitted to the hospital for an expert opinion and transition to supportive care or hospice. A consulting service that is available to see such patients in the ED would make this process more satisfactory.

Therapeutic procedures frequently utilized in cancer patients necessitate the development of certain services. Oncologic patients have a frequent need for invasive procedures such as thoracentesis, paracentesis, stenting, and percutaneous drainage. Some of these procedures can be done by ED physicians, but they are time-consuming and difficult to perform in a busy ED. Several institutions have dedicated teams to help facilitate these procedures. Another common diagnosis is the incidental finding of pulmonary embolus on CT scans. Many of these patients are handled in the emergency centers at Memorial Sloan Kettering, MD Anderson, and OSUWMC.

The optimal medical management of many cancer-related emergencies is an excellent area for further research. Many practice patterns are based on expert opinion or prior experience rather than clinical trials. Formal training for treatment of oncologic emergencies is still up and coming, though models do exist [12]. Otherwise, this skill set currently must be learned through work experience. Examples of frequently treated problems that could be better supported by research are treatment of hyponatremia and hypercalcemia of malignancy, rescue treatment of chemotherapy- or radiationinduced nausea and vomiting, chemotherapyor radiation-induced diarrhea and mucositis, chemotherapyinduced peripheral neuropathic pain, pain related to colonystimulating growth factors, dosage of steroids and radiation in malignant spinal cord compression, and acute management of narcotic-induced constipation. Other important areas include treatment of therapy-associated skin rashes and management of medical problems with unique complications, such as venous thromboembolism and acute coronary syndrome in thrombocytopenic patients and anticoagulation of patients who have metastatic disease to the brain.

In summary, the care model used for patients with oncologic emergencies must be tailored to the local medical and oncologic environment; therefore, it naturally follows that different medical systems have developed different processes to care for these patients. A constant among the models discussed here is the underlying goal of care being provided to these patients by clinicians who are knowledgeable about their needs and have integrated communication with the primary oncologists. Acute care of the oncologic patient is gaining recognition as an important area that could be improved upon with increased training, research, and emphasis on integration into the oncologic system.

References

- National Cancer institute: SEER Database. https://seer.cancer.gov/ statfacts/html/all.html. Accessed 13 Jul 2020.
- Rodriguez MA, Walters RS, Burke TW, editors. 60 years of survival outcomes at the University of Texas MD Anderson Cancer Center. New York: Springer-Verlag; 2013.
- Bluethmann SM, Mariotto AB, Rowland JH. Anticipating the "silver tsunami": prevalence trajectories and comorbidity burden among older cancer survivors in the United States. Cancer Epidemiol Biomark Prev. 2016;25(7):1029–36.
- Rivera DR, Gallicchio L, Brown J, et al. Trends in adult cancerrelated emergency department utilization: an analysis of data from the nationwide emergency department sample. JAMA Oncol. 2017;3(10):e172450.
- Epner DE, Ravi V, Baile WF. When patients and families feel abandoned. Support Care Cancer. 2011;19(11):1713–7.
- Elsayem AF, Bruera E, Valentine AD, et al. Delirium frequency among advanced cancer patients presenting to an emergency department: a prospective, randomized, observational study. Cancer. 2016;122(18):2918–24.
- Ahn S, Lee YS, Lim KS, Lee JL. Emergency department cancer unit and management of oncologic emergencies: experience in Asan Medical Center. Support Care Cancer. 2012;20(9):2205–10.
- Elsayem AF, Gonzalez CE, Yeung S-C, Merriman KW, Todd KH. In-hospital mortality of patients admitted through the emergency department of a comprehensive cancer center. J Clin Oncol. 2012;30(15 Suppl):6125. https://doi.org/10.1200/ jco.2012.30.15_suppl.6125.
- 9. Yeung S-C, Escalante CP. Oncologic emergencies. Hamilton: BC Decker; 2002.
- Rowe BH, Guo X, Villa-Roel C, Schull M, Holroyd B, Bullard M, et al. The role of triage liaison physicians on mitigating overcrowding in emergency departments: a systematic review. Acad Emerg Med. 2011;18(2):111–20.
- 11. Taplitz RA, Kennedy EB, Bow EJ, Crews J, Gleason C, Hawley DK, Langston AA, et al. Outpatient management of fever and neutropenia in adults treated for malignancy: American Society of Clinical Oncology and Infectious Diseases Society of America Clinical Practice Guideline Update. J Clin Oncol. 2018;36(14):1443–53.
- Purcell MG, El Majzoub I. The oncologic emergency medicine fellowship. Emerg Med Clin North Am. 2018;36(3):637–43.

Informatics

James P. Killeen, Christian J. Dameff, Jeremy S. Bass, Brian Khan, Ammar Mandvi, and Nathan Yung

Introduction

Clinical informatics is the study of information technology as it applies to clinical care within the health system. The American Medical Informatics Association (AMIA) considers informatics when used for healthcare delivery to be essentially the same regardless of the health professional group involved. Clinical informatics is concerned with information use in healthcare by clinicians. Clinical informatics includes a wide range of topics ranging from clinical decision support (CDS) to visual images; from clinical documentation to provider order entry systems; and from system design to system implementation and adoption [1]. In this chapter, our goal is to introduce the reader to new and old concepts that will allow the user to assess information and knowledge to meet the needs of healthcare professionals and patients. The reader will be able to characterize and evaluate information technology, so that they are better able to refine clinical workflow processes, develop new processes, implement those processes, and refine clinical decision support systems [2]. Knowledge of these elements will aid providers in clinical care to their patients. We will discuss workflow,

J. P. Killeen $(\boxtimes) \cdot C. J.$ Dameff

Department of Emergency Medicine, Department of Biomedical Informatics, University of California San Diego, San Diego, CA, USA e-mail: jkilleen@ucsd.edu

J. S. Bass Department of Psychiatry, Clinical Informatics, University of California San Diego, San Diego, CA, USA

B. Khan · N. Yung Department of Medicine, Clinical Informatics, University of California San Diego, San Diego, CA, USA

A. Mandvi

Department of Family Medicine, Clinical Informatics, University of California San Diego, San Diego, CA, USA clinical decision support, information technology systems, and communication, concluding with a discussion of cancer registries and research.

Workflow Process Redesign and Quality Improvement

Workflow has been studied both as a concept and a phenomenon. As a concept, workflow is defined as the sequence of physical and mental tasks performed by people within and between work environments. The flow of information, objects, and people using information and objects through space and time represents the phenomena of workflow. Clinical workflow studies aim to model a simplified version of work in the complex healthcare setting [3]. The simplification achieved by modeling aids in making complex systems more comprehensible as a result of the explanatory nature of such models [4, 5].

Multilevel perspectives are useful to understand workflow comprehensively in the complex system of healthcare [6]. Workflow can occur sequentially or simultaneously and at various levels (individual or organization). Workflow occurs interorganizationally, between clinic employees, and for individual employees before, during, and after a patient encounter. Cognitive workflow occurs as cerebral processes in collecting data and making decisions.

An example of workflow can be illustrated in the ordering of a medication. The workflow of ordering a medication includes communication between the provider and the patient, the provider's mental processes, and the physical action by the provider of writing the prescription on paper or electronically into an electronic health record and sending the order electronically or the patient taking the paper prescription to the pharmacy. In this example, one can see the use of cognitive, individual, organizational, and intraorganizational workflow.



Any time there is a change in practice, particularly related to health information technology, workflow changes occur. Delays in patient care, billing, and communication are prone to occur if the workflow is unaccounted for, overlooked, or oversimplified. All healthcare organizations, regardless of size, must identify a person or group to monitor and assess current and anticipated workflow. Workflow information should be collected as early as possible, ideally before implementing a health IT system, and continually assessed including post implementation as a form of continuous process improvement.

Workflow Analysis

Workflow analysis may be used to improve the outcome of healthcare processes and products, including the practice of healthcare informatics. Institute of Medicine landmark reports call for the use of workflow analysis in an effort to improve healthcare quality, efficiency, effectiveness, and safety [7]. Analysis of workflow requires a reduction of a complex process into analyzable parts in a stepwise fashion.

Various tools can be used during workflow analysis, and a single approach will likely capture a small subset of the complexity. Methods to capture workflow data include qualitative, quantitative, and mixed methods [8]. Qualitative methods focus on naturalistic observation of subjects and activities using artifact collection, spatial analysis, and interviews. Quantitative methods are a more structured approach. Time-motion studies can track the efficiency and quality of healthcare workflow, quantifying the time involved in tasks by observation, self-reporting, or automation to collect temporal data [9]. Questionnaires and surveys are also used as workflow analysis methods. Data collected from the electronic medical record including audit logs, a form of metadata, is a new and emerging area of data collection for workflow methodology.

Visualizing workflow is an important tool as it provides users with cognitive support for visualizing detailed processes, showing parallel processes and allowing different perceptions of processes [10]. The most common method of visualization is flowcharting (process mapping). Flowcharting shows how processes really happen, rather than how they are expected or supposed to happen. This method helps one understand what contributes to different types of flows for the same process, find ways to improve the flows, and identify ways that health IT will affect workflows. Flowcharting is accomplished in five general steps: (1) decide on the process to examine; (2) create a preliminary flowchart; (3) add detail to the flowchart; (4) determine who needs to be observed and interviewed; and (5) do the observations and interviews [11]. An example of a flowchart from a patient being diagnosed with cancer and undergoing treatment and follow-up is shown in Fig. 3.1.

Workflow Redesign

The goal of workflow redesign is to create workflow that supports improved outcome of workflow activities (patient care). Workflow reengineering requires deliberate steps including changes to the mental and physical steps of people who move through a workflow process and changes to the steps in the interactions among organizations involved in a process. Karsh and Alper suggested a system to ten steps of process redesign as seen in Table 3.1 [12]. Broadly, process redesign is achieved by assessing the current state, envisioning the desired future state, planning to get to the future state, carrying it out, and evaluating the outcome.

Quality Improvement

Quality improvement in healthcare is a continuous method for improving process performance. Several quality improvement methodologies are used in healthcare. The Plan-Do-Study-Act (PDSA) is a prominent method that leads quality improvement cycles. The "plan" phase includes identifying a problem and potential solutions. "Do" involves a polar testing of a solution. The "study" phase evaluates if the change was successful. "Act" involves adopting, adjusting, or abandoning the implemented solution. Lean is another process improvement strategy that emphasizes value to customers by utilizing root cause analysis to eliminate waste and improve process flow. Six Sigma is another process improvement methodology that emphasizes quantitative and statistical approaches in continuous quality improvement at the project level to reduce process variations and eliminate defects.

Conclusion

Workflow is the sequence of tasks performed by various people within and between work environments. Workflow analysis is an integral part of quality improvement implementation and health informatics. In this chapter, we have outlined workflow analysis tools, a framework for workflow redesign, and gave an overview of quality improvement methodologies.

3 Informatics



Fig. 3.1 Example flowchart of a patient being diagnosed with cancer, undergoing treatment and follow-up

Table 3.1 Karsh and Alper's ten steps of process redesign [12]

Ten step	Ten steps of process redesign							
Step 1	Decide what system will be the subject of analysis							
Step 2	Produce a preliminary work system map							
Step 3	Use the preliminary workflow map to determine who should be represented on the team that will carry out the analysis							
Step 4	Conduct an initial scan of the system with the team							
Step 5	Determine boundaries of the system under study							
Step 6	Determine performance expectations for each step in the system							
Step 7	Begin formal data collection to review and update the workflow map. Gauge the current performance of the system, and determine baseline measures that will be used to evaluate the effectiveness of the redesign							
Step 8	Analyze the data							
Step 9	Once hazards (i.e., causes of failure modes or variance) have been identified, control strategies should be developed							
Step 10	Analyze redesign ideas. Decide on a redesign idea, pilot testing, and implementation							

Clinical Decision Support

The Office of the National Coordinator for Health Information Technology (ONC) defines clinical decision support (CDS) as providing clinicians, staff, patients, or other individuals with knowledge and person-specific information, intelligently filtered or presented at appropriate times, to enhance health and healthcare [13]. CDS are set of tools and logic to assist providers in making uncertain decisions. All medical decisions come with some percentage of uncertainty: diagnosis, testing, natural progression of disease process, treatment, and subsequent effects. CDS has evolved to remove some of the cognitive burden involved in medical decision-making.

There are some fundamental concepts we will review in order to leverage additional tools to aid in CDS. The first is the concept of expected value and expected utility. *Expected value* (mathematical expectation, mean or average) is the random variable in a simplification of the weighted average and intuitively is the arithmetic mean of many independent realizations of that variable [14], whereas *expected utility* concerns people's preferences about choices that have uncertain outcomes (gambles). The expected utility states that the subjective value associated with an individual's gamble is the statistical expectation of that individual's valuations of the outcomes of that gamble, where these valuations may differ from the dollar value of those outcomes [15]. *Expected utility theory* is a theoretical approach to making optimal decisions under risk [16].

An example of these two concepts: in the presence of risky outcomes, a decision-maker does not always choose the option with higher expected value investment. Suppose there is a choice between a guaranteed payment of \$1.00 and a gamble in which the probability of getting a \$100 payment is 1 in 80 chances and the alternative, far more likely outcome (79 out of 80) is receiving \$0. The expected value of the first alternative is \$1.00 and the expected value of the second alternative is \$1.25. According to expected value, people should choose the \$100-or-nothing gamble; however, as stressed by expected utility, some people are risk averse enough to prefer the sure thing, despite its lower expected value. People with less risk aversion would choose the risk-ier, higher-expected-value gamble [15].

Expected value of gambling:

- If you gamble and win, you get \$100.00.
- If you gamble and lose, you get nothing (\$0.00).
- If you don't gamble, you are guaranteed \$1.00.

Formula:
$$100 * \left(\frac{1}{80}\right) + 0 * \left(\frac{79}{80}\right) = 1.25$$

Similar to engineering as it relates to healthcare, diagnostic inferences models have two elements: *tests* and *conclusions*. *Tests* include any source of information that can be used to determine the health of a system. *Conclusions* typically represent faults, including hardware fault modes, functional failures, specific non-hardware failures, and specific multiple failures. A conclusion may also indicate the absence of a failure indication (no fault). With this model, one can revise and refine opinions with imperfect information, comparable to a differential diagnosis. There are three characteristics to consider in making a diagnosis: detection, localization, and isolation, as defined in Table 3.2. In developing a diagnosis, the reader should focus on concepts emphasizing a structured approach to system testing and diagnosis. These include:

Table 3.2 Characteristics considered when making diagnoses

Characteristics	Definition
Detection	The ability of a diagnostic strategy to identify that a failure in some system has occurred
Localization	The ability to say that a fault has been restricted to some subset of the possible causes
Isolation	The identification of a specific fault through some test, combination of tests, or diagnostic strategy

- Maximizing reuse of design and test data, information, knowledge, and software
- Integrating support equipment and manual testing, to provide complete coverage of diagnostic requirements
- Integrating available diagnostic information, to minimize required resources and optimize performance

Capturing the relationships between tests and diagnosis provides a knowledge representation that can be processed by a reasoning system for health management. Initially, equal quality among test results is assumed and that every test outcome reflects the state of the unit being tested. In practice, this assumption is often relaxed to allow a measure of confidence to be associated with each test [17, 18].

Concepts of CDS include *heuristics*, which are patterns of bias in CDS. Heuristics is any approach to problem solving that employs a practical method that is not guaranteed to be optimal, perfect, or rational, but is nevertheless sufficient for reaching an immediate, short-term goal. Heuristics can be mental shortcuts that ease the cognitive load of making a decision [19]. Heuristics are the strategies derived from previous experiences with similar problems. These strategies depend on using readily accessible, though loosely applicable, information to control problem solving in people, machines, and abstract issues [20]. Some of the more common heuristics that apply to healthcare can be seen in Table 3.3 [21–27].

Cost-effectiveness analysis is a form of analysis that compares the relative costs and outcomes (effects) of different courses of action. Cost-effectiveness analysis is distinct from cost-benefit analysis, which assigns a value to the measure of effect. Typically, the cost-effectiveness analysis is expressed in terms of a ratio where the denominator is a gain in health from a measure (years of life, sight-years gained) and the numerator is the cost associated with the health gain. Costutility analysis can be used in decision analysis to define the "value" of an outcome node by adjusting the value of the outcome based on the perceived utility of that outcome for the patient. The most familiar outcome measurement is quality-adjusted life years (QALY) [28].

 Table 3.3
 Common heuristics in healthcare

Heuristics	Definition	Example
Availability	Overestimating the probability of unusual events because of recent or memorable instances [21]	The last patient I saw with symptom X had disease Y, so we should test for Y
Representativeness	Overestimating of a rare disease by matching patients to "typical picture" of that disease [22]	He has features of the rare disease X, so we should test for it
Anchoring	The failure to adjust probability of a disease or outcome based on new information, like "premature closure" [23]	I was told in sign out that he had condition X, so I didn't consider it might be condition Y, despite lab results
Value-induced bias	Overestimating the probability of an outcome based on value associated with that outcome [24]	It would be horrible to miss a brain tumor in this patient with new onset headache, so we should get a head CT
Affect heuristic	A mental shortcut that uses emotion to influence the decision. Emotion is the affect that plays the lead role that makes the decision or solves the problem quickly or efficiently. It may be used while judging the risks and benefits of something [25]	Your "gut decision" about the presentation of a patient
Familiarity heuristic	A mental shortcut applied to various situations in which individuals assume that the circumstances underlying the past behavior still hold true for the present situation and that the past behavior thus can be correctly applied to the new situation [26]	I am familiar and comfortable with the Arrow Triple Lumen kit by Teleflex; I now need an arterial line kit, so I will choose the Teleflex brand since I am familiar with their other products
Simulation heuristic	A simplified mental shortcut in which people determine the likelihood of an event happening based on how easy it is to mentally picture the event happening [27]	When the provider can easily "mentally undo" the sequence of events that led to a specific outcome like the placement of a chest tube or a cardiac arrest resuscitation

So, what makes a good test? Most would say a test with a high sensitivity and high specificity. Sensitivity is the measure of the proportion of actual positives that are correctly identified. Specificity is the measure of the proportion of actual negatives that are correctly identified. Sensitivity is the extent to which actual positives are not overlooked (minimizing false negatives), and specificity is the extent to which actual negatives are classified as such (minimizing false positives). The positive predictive value (PPV) and negative predictive value (NPV) describe the performance of a diagnostic test. A high result can be interpreted as indicating the accuracy of such a test [29]. The false-positive rate is the proportion of all negatives that still yield positive test outcomes, i.e., the conditional probability of a positive test result given an event that was not present. False-positive rate is equal to the significance level. The specificity of the test is equal to 1 minus the false-positive rate. False-negative rate is the proportion of positives yielding negative test outcomes with the test, i.e., the conditional probability of a negative test result given that the condition being looked for is present.

Of note, false positives should be differentiated from the phenomenon of *overdiagnosis* [30]. The finding of an insignificant pulmonary nodule or an adrenal "incidentaloma" on a chest CT ordered for a patient with a suspected pulmonary embolism is an example of overdiagnosis. The use of CDS tools has the potential to minimize, or at least standardize, the use of advanced imaging technology in such cases.

By reviewing the 2×2 tables shown in Table 3.4, we can design the most efficient CDS questions or tests.

Key elements of CDS are best described in a quote from Wyatt and Spiegelhalter: "Active knowledge systems which use two or more items of patient data to generate case-specific advice" [31]. More specifically, leveraging a good foundational knowledge base along with patient-specific information such as vitals or laboratory results and using the most appropriate mode of communication will assist the user to make the most appropriate choice. As the user designs and builds their CDS, it is important to consider the following targets:

- 1. What are the desired outcomes/clinical targets of CDS?
- 2. How will the CDS tool improve efficiency?
- 3. Are we looking for early detection/screening of the CDS?
- 4. Can CDS assist in the diagnosis or treatment protocol?
- 5. Can CDS provide preventative adverse outcome?
- 6. Can CDS provide follow-up management?
- 7. How does CDS provide cost reductions/conveniences?

Other design considerations should include the target audience. Which member of the healthcare team is the target for CDS? Is the intervention targeted to patients or families? Also consider the level of control of the CDS (preemptive, suppressible, hard-stop, or interruptive). Preemptive or active CDS is a rule based upon simple logic or systems-based upon probability. Active CDS includes rules and alerts. Respectively, hard-stop or suppressible control levels either prevent the user from taking an action altogether or allow them to proceed only with the external override of a third party. Interruptive CDS occurs when a process is interrupted

Table 3.4 Table to derive sensitivity and specificity

Total Population	Condition Positive	Condition Negative	Prevalence			
Predicted Condition Positive	True positive	False positive	Positive Predictive Value True Positives True Positives+False Positives			
Predicted Condition Negative	False Negative	True Negative	Negative Predictive Value True Negatives False Negatives+True Negatives			
	Sensitivity <u>True Positive</u> True Positives+False Negatives	Specificity <u>True Negative</u> True Negatives+False Positives				
	False Negative Rate False Negatives True Positives+False Negatives	True Negative Rate True Negatives False Positives+True Negatives				
True positive	Sick	people correctly identified as	sick			
False positive	Health	y people incorrectly identified	as sick			
True negative	Healthy	people correctly identified as	healthy			
False negative	Sick people incorrectly identified as healthy					

Table 3.5 Example categories of clinical decision supports

Therapeutic duplication Single and cumulative dose limits

Allergies and cross allergies

- Contraindicated route of administration
- Drug-drug and drug-food interactions

Duplicate orders

Contraindications/dose limits based on patient diagnosis, age,

weight, prior laboratory, or radiology studies

and requires the user to acknowledge its information by taking one or more actions, such as in computerized order entry (CPOE) systems. Three types of interruptiveness are ondemand (link to formulary from within order), in-line or modeless (unread lab result notification on sidebar), and popup or modal (alerts or reminders requiring acknowledgment).

Table 3.5 shows some examples of CDS categories.

When designing CDS, the user should always ask themselves the following questions to make sure they have addressed the five "rights" to assess their success [32]:

- 1. Right information quality of knowledge base
- 2. Right person target of CDS
- 3. Right format implementation of CDS (speed, ease of use, comprehensibility)

- 4. Right channel mode of CDS
- 5. Right time workflow integration

Do I have the right information for the question? Have I accessed the right knowledge base and provided the correct resources and references? Who is my target audience, and have I reached them successfully? Do I have my question in the right format? Am I providing them with knowledge only, or is my aim to stop the user's process or redirect them? Do I have my CDS in the correct spot to provide the user the correct additional knowledge to make an informed decision?

Having created a CDS plan or outline, the user will most likely need to submit a proposal to a CDS committee that oversees all CDS and provides continuous feedback for the system. Many institutions may have forms to complete or submit. You will see in Fig. 3.2 that the example CDS form request follows the "5 Rights."

David Bates summarized the goals and expectations for CDS in his 2003 AMIA article (Table 3.6) [33]. He believes it is key that information systems provide decision support to users at the time they make decisions, thus promoting improved quality of care. Providers make many errors, and clinical decision support should help identify and avoid such errors.

Clinical Decision Support Request Form

Person submitting the request	
Purpose of goal for the CDS	
Evidence-based need	RIGHT information What are the clinical grounds and supporting evidence for the CDS?
Intended audience	RIGHT person Who will see it? - e.g., physicians, RNs, pharmacists, etc.
Clinical champion(s)	Who will be the clinical champion (s) responsible for the success of implementing this alert?
Primary stakeholders	Who are the primary group (s) or committee (s) that have a stake in this CDS? This is important, as they may need to sign off prior to enabling CDS.
Established level of consensus?	Do we know whether stakeholders all agree on need for CDS?
Clinical criteria for the CDS	RIGHT time – workflow integration When should the CDS be triggered?
Impact on patienr care	What is expected to change as a result of the CDS?
Impact on workflows	RIGHT channel – mode of CDS What groups, activities and workflows is the CDS limited to? What is the frequency that it will occur?
Urgency	What is the urgency for this CDS? Why is it urgent?
Monitoring success of the CDS	RIGHT format – implementation of CDS (speed, ease of use, comprehensibility) <i>What specific outcomes will be used to determine</i> <i>success of the CDS</i> ?
Future clinical review	What is the expected frequency and method for future clinical review of the CDS?

Fig. 3.2 Example of a clinical decision support (CDS) form request that follows the "5 Rights"

Table 3.6 Ten commandments for effective clinical decision support (33)

- 1. Speed is everything expect sub-second latency
- 2. Anticipate needs and deliver in real time -e.g., showing relevant labs with med orders
- 3. Fit into users' workflow external tools are not as good as those at POC
- 4. Little things can make a big difference "usability matters a lot," "make it easy to do the right thing"
- 5. Physicians resist stopping do not tell doctors to not do something without offering an alternative
- 6. Changing direction is easier than stopping
- 7. Simple interventions work best try to fit guidelines onto a single screen
- Asked for additional information when you really need it – "likelihood of success is inversely proportional to the number of extra data elements needed"
- 9. Monitor impact, get feedback, and respond. Evaluate your CDS
- 10. Manage and maintain your knowledge-based systems. Keep up with clinical care

Information Technology Systems

Telehealth

The desire to constantly improve the access, delivery, and quality of healthcare has resulted in the application of novel technologies to nearly all domains of medicine. Telehealth, also referred to as telemedicine, is defined by CMS as the electronic transmission of patient information from one distant site to another and has evolved to also include electronic communication between patients and providers in order to facilitate healthcare. Telehealth can employ many different types of technology to achieve patient to clinician communication including telephonic, short message service (SMS), fax, email, and real-time audio-video communication utilizing Internet connectivity and computers.

The utility of telehealth is vast and constantly evolving. Commonly cited benefits include improved clinician access in rural areas where such expertise is not available, decreased healthcare costs, and increased healthcare system workforce resilience, as well as patient and provider convenience [34]. During the COVID-19 pandemic, telehealth was utilized as a mechanism to reduce the use of one-time use personal protective equipment [35]. Some patients report preferring telehealth in place of traditional in-person visits for certain encounters, such as for birth control prescriptions [36]. Immunocompromised patients, including oncologic and rheumatology patients, have utilized telehealth to limit pathogen exposure associated with in-person visits at healthcare facilities.

Case Study 1

A 65-year-old female with a history of acute promyelocytic leukemia (APL) presents by ambulance to a rural hospital emergency department (ED) with complaints of altered mental status and worsening rash. Her husband at bedside reports 1 day of fever, increasing confusion, and, most recently, epistaxis. On examination of her skin, a diffuse petechial rash is discovered.

The clinician at bedside recognizes the patient is likely in disseminated intravascular coagulation (DIC) likely due to her APL. Basic supportive interventions are initiated, and diagnostic testing is ordered to confirm the suspected diagnosis. The clinician recognizes that this patient would benefit from specialty consultation and transfer to a higher level of care. A tablet computer mounted on a bedside cart is then used to connect a live video communication with an oncologist hundreds of miles away. After consultation with the oncologist, a live encounter with the patient and spouse is performed by the remote physician. Written treatment recommendations are captured and electronically transmitted by the electronic health record (EHR) from the oncologist to the emergency physician, and transfer to a quaternary care center is initiated.

History of Telehealth Although telehealth is widely used today all across the globe, its origins began over 60 years ago with the application of live video to facilitate psychiatric evaluations [37]. The National Aeronautics and Space Administration (NASA) researched telehealth heavily throughout the 1960s–1970s, culminating in pilot studies delivering healthcare to astronauts during space travel [38]. The birth of the Internet in the early 1990s, coupled with increased use of consumer video conferencing applications such as Skype in the early 2000s, further advanced the adoption of real-time audio-video communications as a way to deliver healthcare.

Today, there are hundreds of software and hardware telehealth products and services. Many telehealth encounters are performed on smartphones or tablet computers, as such devices have become ubiquitous in many parts of the world. The integration of telehealth with remote biometric sensor technology has also created new opportunities, such as telehealth intensive care units (ICU) and remote stroke consultation. Although many clinicians today do not incorporate telehealth into their practice, some have elected to only practice by telehealth, often citing increased flexibility and consumer demand.

Technology The safe evaluation and care of patients in any environment or by any medium requires some basic standards in place, such as clear audio, adequate visuals, and timely data access. These specific standards initially led to the development of specialized telehealth hardware such as high-definition cameras and microphones. Early hardware was often large, bulky, fragile, and expensive. Early telehealth efforts were hindered by lagging pixelated video and fragmented audio. As Internet bandwidth increased and wireless technology decreased barriers to bringing devices to patients' bedsides, older hardware was replaced with smaller, cheaper, and better-quality devices, often affixed to carts.

Along with advances in hardware came new specialized healthcare software: the electronic health record (EHR). The replacement of paper medical records by digital records further aided telehealth adoption, as patient records could be stored on servers or in the cloud to be accessed by clinicians quickly and remotely. Most modern EHRs also support computerized provider order entry (CPOE) allowing rapid electronic ordering and reporting of diagnostic tests and ordering of medication prescriptions. Some EHR vendors have now started building telehealth directly into their products.

Telehealth Regulation Telehealth in the United States, like much of healthcare generally, is highly regulated at both state and federal levels. Restrictions on telehealth traditionally fall under three main categories: allowable locations/ applications, requirements of service, and billing. Rural patients without access to robust care were seen as the greatest beneficiaries of telehealth. This led to regulations limiting telehealth services in urban areas where the need was postulated to be less. Furthermore, limits on what types of encounters (ambulatory vs. inpatient) qualify for telehealth are commonplace. The requirements of service are also used to tier the classification for reimbursement of telehealth. For example, a simple telephone audio-only call may not qualify as telehealth despite robust history gathering. Regardless of what type of telehealth tier is being utilized, the patient must verbally consent to telemedicine services. The business viability of telehealth has heavily relied on reimbursement policy set by agencies such as the Centers for Medicare & Medicaid Services (CMS). Many telehealth services are not reimbursed at the same rates as corresponding in-person encounters.

Many federal and state telehealth regulations were revisited and changed to respond to the COVID-19 pandemic of 2020. This led to the single greatest expansion of telehealth services in history and ushered in a new era where telehealth is commonplace and widely accepted. One of the most notable changes regarded the enforcement of patient privacy and security requirements from the 2009 Health Insurance Portability and Accountability Act (HIPAA). Although HIPAA still applies broadly, the enforcement of high encryption standards in video conferencing software was deferred. This allowed free consumer-focused telecommunication applications such as Skype (Microsoft Inc., Redmond Washington), Zoom (Zoom, San Jose California), and Hangouts (Google, Mountain View California) to be used instead of expensive niche telemedicine platforms.

Oncologic Emergencies and Telehealth The use of telehealth in the prevention of oncologic emergencies is growing. Telehealth may be deployed to reduce exposure to infectious disease in cancer patients at higher risk (e.g., immunocompromised), and telehealth has the obvious potential to reduce ED visits among cancer patients. Telehealth can also facilitate continuity of clinician-patient care, bringing in valuable context from the patient's own specialist when traveling. Remote video consultation and evaluation by oncologic specialists in both regular and emergent capacities can also aid in the diagnosis and treatment of patients. Many rural locations lack robust medical care systems and specialist oncologic expertise can be scant. Much more research is needed on the impact of telehealth on patient outcomes.

Conclusion Telehealth provides powerful tools in the care of oncologic patients during emergencies. Rapid technological developments will continue to change how clinicians care for patients. Changes in the regulation of telehealth have greatly expanded its applications and viability as a regular component of healthcare in the twenty-first century. Leveraging telehealth to assist in the care of patients with oncologic emergencies will prove more common and more powerful in the years to come.

Security

As healthcare continues to innovate and advance, the use of technology to care for oncologic patients continues to evolve and grow. From Internet-connected medical devices to artificial intelligence and machine learning, healthcare is increasingly digitized, connected, and complex. In this era of hyper-connected healthcare, it is important to focus not only on the care of oncologic patients but also on cybersecurity and privacy of sensitive patient data.

Case Study 2

After a long and busy shift in the ED, an attending physician posts on social media the following statement: "Just had the honor to treat one of our nation's last surviving World War II veterans in the Emergency Department at General Hospital! Despite his chronic lymphocytic leukemia, Albert is going to be okay! #VeteransRock #CancerSucks." The next day, he is Table 3.7 18 HIPPA patient identifiers Name Address (all geographic subdivision smaller than state, including street address, city-county, and zip code) All elements (except years) of dates related to an individual (including birthdate, admission date, discharge date, date of death, and exact age if over 89) Telephone numbers Fax number Email address Social Security number Medical record number Health plan beneficiary number Account number Certificate or license number Vehicle identifiers and serial numbers, including license plate numbers Device identifiers and serial numbers Web URL Internet Protocol address Finger or voiceprint Photographic image - photographic images are not limited to images of the face Any other characteristic that could uniquely identify the individual

called into an administrator's office and is subsequently terminated from his employment for violating the hospital privacy policy.

Health Insurance Portability and Accountability Act (HIPAA) The Health Insurance Portability and Accountability Act (HIPAA) was a law passed by the US Congress in 1996 that, among other things, provided regulation around the security of protected health information (PHI) [39]. Beyond defining PHI, this law provided 18 "identifiers" (Table 3.7) constituting sensitive data elements that can be used to identify and subsequently violate the privacy of patients. Additionally, the law established a reporting and enforcement mechanism to ensure parties responsible for protecting PHI could be heavily fined if they suffered a breach or were negligent in securing the data.

Today, HIPAA continues to be a very important part of healthcare regulation. It remains regularly enforced, leading many hospitals to devote significant resources to the protection of PHI and compliance with federal regulation. When PHI is lost or exposed, it is termed a breach. Breaches of greater than 500 patient records often result in mandatory reporting to the federal government as well as the patients whose records were compromised. Common causes of breaches include (1) failure to dispose of paper records properly, (2) loss of computers containing PHI, (3) hacking of records by malicious actors, and (4) loss or records by a third party that was trusted with records (e.g., healthcare contractor or business affiliate). **Conclusion** In the care of oncologic patients, as with other patients, the security and privacy of their PHI is important. As healthcare continues to become more digitized, the risk of exposing this information increases. Federal law protects patient data privacy, and failure to protect these data can lead to significant harms to patients, providers, and organizations. Oncologic patients can be at particular risk of PHI breach as they are often high utilizers of healthcare resulting in more records and can carry sensitive diagnoses.

Communication

Order Sets

AMIA defines an *order set* as a predefined template. Order sets are lists of orders that frequently include medication, laboratory, nursing, diet, activity, and other orders. They existed prior to the advent of electronic health records as paper templates. A common example is an admission order set. This would frequently include an admission order, a diet order, nursing orders, vital signs, activity orders, IV orders, medication orders, laboratory orders, radiology orders, consultation orders, and provider preferences. Order sets allow physician to easily select from commonly used orders to save time and ensure consistency for certain procedures, such as a surgery, admission, or discharge [40].

AMIA indicates that order sets have been "...the standard of care in hospitals for many years. While in the past, it took the form of pen and paper, today, it is, indeed, electronic" [41]. The Institute for Safe Medication Practices (ISMP) has developed guidelines around order sets [42]. The ISMP indicates that well-designed order sets have the potential to "integrate and coordinate care by communicating best practices through multiple disciplines, levels of care, and services, modify practice through evidence-based care, reduce variation and unintentional oversight through standardized formatting and clear presentation of orders, enhance workflow with pertinent instructions that are easily understood, intuitively organized, and suitable for direct application to current information-management systems and drugadministration devices, decrease the potential for medication errors through integrated safety alerts and reminders, and reduce unnecessary calls to prescribers for clarifications and questions about orders" [42]. The ISMP goes on to state that order sets that "are not carefully designed, reviewed, and maintained to reflect best practices and ensure clear communication, they may actually contribute to errors" [42].

The astute observer will note that order sets have been used for many years to standardize workflows, remind providers, and make suggestions about clinical care that has been vetted by best practices and evidence. In the case of patients presenting to the ED, oncologic patients present a unique challenge. These patients are frequently immunosuppressed, which leaves them susceptible to several unique conditions, such as those infections that only spread in the immunocompromised state (i.e., neutropenic fever) and that may present with various metabolic derangements, such as tumor lysis syndrome. Given the infrequency with which emergency physicians encounter these conditions and the morbidity and mortality associated with them, these cases are ripe for use of order sets. Oncologic emergencies also demonstrate the need for collaboration in design of order sets. Many large healthcare organizations have informatics teams of healthcare practitioners that work in concert with information systems (IS) personnel to develop content for their electronic health record (EHR). The oncologic emergency is an example where various stakeholders and specialties work together to develop content. Polling providers for preference, along with scouring the literature for recommendations and guidelines, is often the first step in designing an order set. Usually one or more clinical "champions" are identified to begin the process of consulting literature, guidelines, experts in the domain, and practitioners in the affected departments. Their next step is usually to form a working group of affected stakeholders. In this case, ED providers, oncologists, and nursing would likely comprise the group. The two specialties would then discuss recommendations for order sets and request feedback from their respective departments. Much like the legislative reconciliation process, the groups then rejoin, find common ground, and resolve differences. This design would then be submitted to IS for testing and, later, implementation. Once implemented, as indicated by the ISMP, a properly verified and scrutinized order set has the ability to standardize and improve care.

Transition of Care Tools

The order set itself is one way of communicating care standards. The literature on emergency physician to ambulatory provider and vice versa communication is sparse, but it demonstrates differences in communication preferences [43]. Transitions of care are a topic of much discussion and are heavily scrutinized by The Joint Commission [44]. This is especially the case after an ED visit. While not all visits to the ED or hospital are avoidable, there has been increased attention in recent years on avoiding as many visits as possible.

One factor contributing to avoidable ED visits is providerto-provider communication. Open and clear communication decreases errors and costs [45]. Some health systems have found communication to be so important for patient care that caseloads have been limited to ensure that provider-toprovider communication takes place [46]. With all the focus on communication, one might assume this problem would have been resolved. However, communication is regularly cited for the last 15+ years as one of the major factors in malpractice cases, regulatory citations, and poor patient reviews [47].

Healthcare organization management has taken this seriously and imagined a variety of solutions. After the HITECH Act was signed into law, electronic health records (EHRs) adoption greatly increased. It was theorized that EHRs would foster provider-to-provider communication. Communication increased, but it was mostly asynchronous communication (i.e., email, text messages, assigning providers to notes, etc.). This was a different form of communication than existed previously, which was largely sharing information face-toface or via telephone [48]. If these communications methods were equal in terms of patient care, this chapter would end early. Interviewees have indicated that EHRs allow for easier and more frequent asynchronous communication, though this does not remove the need for physician-to-physician communication. Learning from each other is much less likely to occur through an email as opposed to a phone call. Proposed solutions involving the EHR include building infrastructure to allow for "preferred mode of contact" and standardizing communications [48]. EHRs may assist with this, but if the workflow hasn't been designed prior to initiation, it's easy for staff to use more and more asynchronous communication.

In addition, proper configuration of EHRs is necessary to ensure that the right information flows to the right person, in the right format and channel, at the right time. These rights are collectively known as the "5 Rights of Clinical Decision Support" [49]. This is usually referenced regarding tools in the EHR, though the rights apply to any information in an EHR. Globally, it applies to communication in general. The authors recommend a similar approach to that applied for workflow analysis, order set design, and other aspects of informatics. This is to perform a thorough analysis of the situations in question, engage stakeholders and leaders, form consensus, test, implement, and review. The advantage of engaging leaders (or the early adopters) in the department or division is the outsized influence they may have on those resistant of adoption. It equally applies in the design of EHR implementation, design, and flow of information. It is critical to find the consensus on the preferred method of communication. When differences occur, technology may assist to resolve the issue of differing preferences.

At the authors' institution, the EHR was utilized to engage clinicians, administrators, researchers, and stakeholders by considering a patient visit for particular specified reasons as a "unit of communication." For a given patient visit, different stakeholders wished to be notified in different ways. Some clinicians preferred a text notification, while others preferred an email. Administrators preferred a spreadsheet or interactive database. Informatics personnel coordinated with leadership in IS, the ED, ambulatory space, and leadership to develop a system that would create reports and send notifications to a group of providers. Review of the system after implementation demonstrated increased satisfaction for all parties involved. An additional bonus was closer to real-time data on those presenting to the ED for the identified reasons. Before this system, monthly SQL queries were necessary to create spreadsheets and graphs to monitor patient care. The new system utilized EHR tools and allowed those involved in care to easily identify the individuals affected, as well as relevant data to inform care processes.

The agreement on the pre-implementation of this plan led to less utilization of resources to better understand trends in the healthcare system. It also led to improved adherence with notification, as the prior system would not always notify the provider. Prior to implementation, ED staff was to send a secure email to the identified individuals. Adherence with this was poor, as the event was not common enough for the ED providers to implement the event in their workflow. For a variety of technical reasons, an order set for this scenario could not be implemented. However, enhanced communication was fostered utilizing the methods outlined in this chapter.

Research and Registries

Cancer Registries

A cancer registry is an information system that collects and analyzes data from a census of cancer cases. Registry data can be used to define and monitor cancer incidence, investigate treatment patterns, evaluate efforts to prevent cancer, and improve survival [50]. This allows public health officials and healthcare professionals to be apprised of cancer-related measures used to guide cancer prevention and control efforts.

Cancer data are collected in two different types of registries. Population-based registries are tied to state health departments, while hospital registries are a part of a healthcare organization's cancer program [51]. Population-based registries collect information on all cases within a certain geographic area from multiple reporting organizations including hospitals, doctors' offices, nursing homes, clinical labs, and ambulatory care organizations, as well as chemotherapy and radiation treatment centers. Hospital registries provide more complex data used to assess clinical care at a particular hospital. These data typically guide education of healthcare providers and focus on patient care. Pooled data can be used to observe trends with specific populations, providers, or locales [52, 53].

Cancer registrars with standardized training aid in collecting data for cancer registries. Registrars prepare accurate and timely data that is reported to the registry. Identifying individuals with cancer, or *casefinding*, is the first step in cancer registration. This is initiated during clinical care when physicians note the cancer site, type, stage, and patient demographics in the medical record. Registrars summarize and record other information for certain registries, such as treatments and follow-up to record recurrence and survival data [52]. The HITECH Act, through the electronic health record meaningful use program, incentivized case-based reporting.

Cancer Surveillance Programs

In 1971, the National Cancer Act was established which mandated collection, analysis, and sharing of cancer patient data in the United States for research, detection, and treatment of cancer. The National Cancer Institute established the first national cancer registry, the Surveillance, Epidemiology, and End Results (SEER) program, in 1973. This large population-based system of cancer registries provides data on cancer incidence, mortality, treatment, and survival [51, 53]. Data are collected regionally, representing 28% of the US population [51].

In 1992, the US Congress established the Cancer Registries Amendment Act which authorized the Centers for Disease Control and Prevention (CDC) to provide regional assistance to improve cancer registries, implement registries in absent regions, model legislation, provide training for registry personnel, set standards, and aid in establishing a reporting and data processing system. The National Program of Cancer Registries (NPCR) was established to accomplish these goals and supports cancer registries in 45 states, representing 96% of the population [53]. The NPCR and the SEER program, together, collect data for the entire United States.

Together with data regarding cancer incidence and death rates, cancer survival measures provide a comprehensive picture of the burden of cancer in a population and support public health efforts to prevent new cancers, extend survival and quality of life after a cancer diagnosis, and reduce cancer health disparities [50].

References

- Clinical Informatics | AMIA [Internet]. AMIA.org. 2020 [cited 22 Jul 2020]. Available from: https://www.amia.org/ applications-informatics/clinical-informatics.
- Gardner RM, Overhage JM, Steen EB, Munger BS, Holmes JH, Williamson JJ, Detmer DE, AMIA Board of Directors. Core content for the subspecialty of clinical informatics. J Am Med Inform Assoc. 2009;16(2):153–7.
- Nemeth C, Cook R, Woods D. The messy details: insights from the study of technical work in healthcare. IEEE Trans Syst Man Cybern Syst Hum. 2004;34(6):689–92.
- 4. What is workflow? | AHRQ Digital Healthcare Research: informing improvement in care quality, safety, and efficiency [Internet]. Digital.

ahrq.gov. 2020 [cited 22 Jul 2020]. Available from: https://digital.ahrq.gov/health-it-tools-and-resources/evaluation-resources/ workflow-assessment-health-it-toolkit/workflow.

- Franck R. The explanatory power of models: bridging the gap between empirical and theoretical research in the social sciences. Boston: Kluwer Academic; 2002. p. 309.
- 6. Carayon P, Karsh B-T, Cartmill RS, et al. Incorporating health information technology into workflow redesign--summary report. (Prepared by the Center for Quality and Productivity Improvement, University of Wisconsin–Madison, under Contract No. HHSA 290-2008-10036C). AHRQ Publication No. 10-0098-EF. Rockville: Agency for Healthcare Research and Quality. October 2010. Available from: https://digital.ahrq.gov/sites/default/ files/docs/citation/workflowsummaryreport.pdf.
- Institute of Medicine (US) Committee on Quality of Health Care in America. In: Kohn LT, Corrigan JM, Donaldson MS, editors. To err is human: building a safer health system. Washington (DC): National Academies Press (US); 2000.
- Finnell J, Dixon B. Clinical informatics study guide. Cham: Springer International Publishing; 2016.
- Lopetegui M, Yen PY, Lai A, Jeffries J, Embi P, Payne P. Time motion studies in healthcare: what are we talking about? J Biomed Inform. 2014;49:292–9.
- Tory M, Moller T. Human factors in visualization research. IEEE Trans Vis Comput Graph. 2004;10(1):72–84.
- Jordan B. Ethnographic workplace studies and CSCW. In: Shapiro D, Tauber M, Traunmüller R, editors. The design of computer supported cooperative cork and groupware systems. Amsterdam: Elsevier; 1996. p. 37–9.
- 12. Karsh B, Alper S. Work system analysis: the key to understanding health care systems [Internet]. Agency for Healthcare Research and Quality; 2020. Available from: https://www.ahrq.gov/sites/default/ files/wysiwyg/professionals/quality-patient-safety/patient-safetyresources/resources/advances-in-patient-safety/vol2/Karsh.pdf.
- Clinical Decision Support | HealthIT.gov [Internet]. Healthit.gov. 2020 [cited 22 Jul 2020]. Available from: https://www.healthit.gov/ topic/safety/clinical-decision-support.
- Expected value [Internet]. En.wikipedia.org. 2020 [cited 22 Jul 2020]. Available from: https://en.wikipedia.org/wiki/ Expected_value.
- Expected utility hypothesis [Internet]. En.wikipedia.org. 2020 [cited 22 Jul 2020]. Available from: https://en.wikipedia.org/wiki/ Expected_utility_hypothesis.
- Kahneman D, Tversky A. Prospect theory: an analysis of decision under risk. Econometrica. 1979;47(2):263–92.
- Bartolini A, Sheppard JW, Munns TE. An application of diagnostic inference modeling to vehicle health management. 2001 IEEE autotestcon proceedings. IEEE systems readiness technology conference. (Cat. No.01CH37237), Valley Forge; 2001. p. 706–17. https://doi.org/10.1109/AUTEST.2001.949454.
- Burns M, Pearl J. Causal and diagnostic inferences: a comparison of validity. Organ Behav Hum Perform. 1981;28(3):379–94.
- Heuristics Explanation and examples Conceptually [Internet]. Conceptually. 2020 [cited 22 Jul 2020]. Available from: https://conceptually.org/concepts/heuristics.
- 20. Pearl J. Heuristics. Reading: Addison-Wesley; 1985.
- 21. Harvey N. Use of heuristics: insights from forecasting research. Think Reason. 2007;13(1):5–24.
- Kahneman D, Slovic P, Tversky A, editors. Judgment under uncertainty: heuristics and biases. Cambridge: Cambridge University Press; 1982. https://doi.org/10.1017/CBO9780511809477.
- Smith HD. Use of the anchoring and adjustment heuristic by children. Curr Psychol. 1999;18(3):294–300.
- Levy A, Hershey J. Value-induced bias in medical decision making. Med Decis Mak. 2008;28(2):269–76.

- 25. Slovic P, Finucane M, Peters E, MacGregor D. The affect heuristic. Eur J Oper Res. 2007;177(3):1333–52.
- Park C, Lessig V. Familiarity and its impact on consumer decision biases and heuristics. J Consum Res. 1981;8(2):223.
- 27. Kahneman D, Tversky A. Variants of uncertainty. Cognition. 1982;11(2):143–57.
- Neumann P, Sanders G, Russell L, Siegel J, Ganiats T. Costeffectiveness in health and medicine. 2nd ed. Oxford UK: Oxford University Press; 2017.
- Altman D, Bland J. Statistics notes: diagnostic tests 2: predictive values. BMJ. 1994;309(6947):102.
- Brodersen J, Schwartz L, Heneghan C, O'Sullivan J, Aronson J, Woloshin S. Overdiagnosis: what it is and what it isn't. BMJ Evid Based Med. 2018;23(1):1–3.
- Wyatt J, Spiegelhalter D. Field trials of medical decision-aids: potential problems and solutions. Proc Annu Symp Comput Appl Med Care. 1991:3–7.
- 32. Osheroff J, Teich J, Levick D, Saldana L, Velasco F, Sittig D, et al. Improving outcomes with clinical decision support: an implementer's guide. 2nd ed. Chicago: Healthcare Information and Management Systems Society; 2012.
- 33. Bates DW, Kuperman GJ, Wang S, Gandhi T, Kittler A, Volk L, et al. Ten commandments for effective clinical decision support: making the practice of evidence-based medicine a reality. J Am Med Inform Assoc. 2003;10(6):523–30.
- Dorsey E, Topol E. State of telehealth. N Engl J Med. 2016;375(2):154–61.
- Turer RW, Jones I, Rosenbloom ST, Slovis C, Ward MJ. Electronic personal protective equipment: a strategy to protect emergency department providers in the age of COVID-19. J Am Med Inform Assoc. 2020;27(6):967–71.
- 36. Yoost JL, Starcher RW, King-Mallory RA, Hussain N, Hensley CA, Gress TW. The use of telehealth to teach reproductive health to female rural high school students. J Pediatr Adolesc Gynecol. 2017;30(2):193–8.
- Hilty DM, Ferrer DC, Parish MB, Johnston B, Callahan EJ, Yellowlees PM. The effectiveness of telemental health: a 2013 review. Telemed J E Health. 2013;19(6):444–54.
- Freiburger G, Holcomb M, Piper D. The STARPAHC collection: part of an archive of the history of telemedicine. J Telemed Telecare. 2007;13(5):221–3.
- Ness R. Influence of the HIPAA privacy rule on health research. JAMA. 2007;298(18):2164.
- UCLA Medical Center. Doctor's Admission Orders. Heart Failure Admission Orders. UCLA HF Orders 3 ALLL Units-2003.doc. Accessed 14 Oct 2020.
- Cowden D, Barbacioru C, Kahwash E, Saltz J. Order sets utilization in a clinical order entry system. AMIA Annu Symp Proc. 2003;2003:819.
- Grissinger M. Guidelines for standard order sets. P T. 2014;39(1):10–50.
- 43. Rider AC, Kessler CS, Schwarz WW, Schmitz GR, Oh L, Smith MD, Gross EA, et al. Transition of care from the Emergency Department to the outpatient setting: a mixed-methods analysis. West J Emerg Med. 2018;19(2):245–53.
- National Patient Safety Goals [Internet]. Jointcommission.org. 2020 [cited 22 Jul 2020]. Available from: https://www.jointcommission.org/standards/national-patient-safety-goals.
- 45. Leonard M, Graham S, Bonacum D. The human factor: the critical importance of effective teamwork and communication in providing safe care. Qual Saf Health Care. 2004;13(Suppl 1):i85–90.
- 46. Wharton University of Pennsylvania. Knowledge @ Wharton. How the mayo clinic built its reputation as a top hospital [Internet]. 2020 [cited 22 Jul 2020]. Available from: https://knowledge.wharton. upenn.edu/article/mayo-clinics-secret-success/.

- STAT. Health. Bailey M. Communication failures linked to 1,744 patient deaths in five years, US malpractice study finds. 1 Feb 2016. [Internet]. [cited 22 Jul 2020]. Available from: https://www.statnews.com/2016/02/01/communication-failures-malpractice-study/.
- Shannon D. Effective physician-to-physician communication: an essential ingredient for care coordination. Physician Exec. 2012;38(1):16–21.
- Campbell R. The five rights of clinical decision support: CDS tools helpful for meeting meaningful use. J AHIMA. 2013;84(10):42–7. (web version updated Feb 2016).
- White M, Babcock F, Hayes N, Mariotto A, Wong F, Kohler B, et al. The history and use of cancer registry data by public health cancer control programs in the United States. Cancer. 2017;123:4969–76.
- American Cancer Society. Cancer surveillance programs the United States [Internet]. Cancer.org. 2020 [cited 22 Jul 2020]. https://www. cancer.org/cancer/cancer-basics/cancer-surveillance-programsand-registries-in-the-united-states.html.
- National Cancer Registrars Association. Cancer registry profession [Internet]. Ncra-usa.org. 2020 [cited 22 Jul 2020]. https://www. ncra-usa.org/About/Cancer-Registry-Profession.
- 53. Centers for Disease Control and Prevention. National Program of Cancer Registries (NPCR). How Cancer Registries Work. How cancer registries work | CDC [Internet]. Cdc.gov. 2020 [cited 22 Jul 2020]. https://www.cdc.gov/cancer/npcr/value/registries.htm.

Tracy E. Spinks and Michael G. Purcell

Case Study: Addressing Unmet Palliative Care Needs in the ED

The medical director of an emergency department (ED) at a tertiary care center reviewed admission data for his cancer population. Over the past 12 months, he—along with colleagues—noticed a trend toward increasing admission rates in this population. On further evaluation of the primary admission diagnoses, the director found that many patients were being admitted for symptom control and palliative care consults. This equated to increasing admissions with prolonged hospital stays and contributed to consistently high hospital capacity. With the resultant decreased bed availability, the other admitted ED patients had longer wait times to arrive in an inpatient bed. The downstream effects directly contributed to ED boarding, longer ED wait times, and decreased patient satisfaction.

To address the issue, the director pulled together a focus group of providers to discuss the daily challenges surrounding cancer patients admitted for symptom control and palliative consults. It became apparent that often there is both discomfort and unfamiliarity of ED providers when it comes to dosing narcotics and antiemetics in this population as they are frequently on long-acting opioids and high doses of breakthrough medicines at home. The focus group expanded to include a member from the division of palliative care. A joint effort led to the creation of palliative care rounds, where the palliative care service would stop by the ED every morning to check in with care providers and assist with any question on symptom control for currently active patients. Additionally, the palliative care service worked to prioritize consults from the ED to help with early, aggressive symptom

T. E. Spinks (🖂)

M. G. Purcell Department of Emergency Medicine, Ohio State University Wexner Medical Center, Columbus, OH, USA control and goals-of-care discussions. The primary goal was to improve symptom control and earlier palliative care access among cancer patients, as measured by two metrics: (1) hospital admissions among cancer patients for symptom control and palliative care consults and (2) palliative care consults in the ED.

After a 3-month pilot, the director reviewed the data. Admissions for symptom control and palliative care steadily declined over the 3 months, while palliative care consults in the ED steadily increased. Unintended benefits of the program led to improved emergency provider comfort with higher doses of narcotics and long-lasting opioids, improved relationships with the division of palliative care, and improved patient satisfaction with symptom control. The data were shared at the hospital's quality forum and hailed as a huge success.

Introduction/Background

Quality issues in oncologic emergency care are well-known. Common ED concerns include overcrowding, long wait times (perceived and actual), boarding, ambulance diversions, inadequate access to specialists, and patient handoffs. Additionally, some issues (e.g., receiving a late-stage cancer diagnosis in the ED) are not directly related to the quality of emergency care. Instead, they reflect broader cancer quality issues, such as inadequate access to and utilization of cancer prevention and diagnostic services, insufficient care coordination, fragmented healthcare delivery, poor symptom management, and underutilized hospice and palliative care services.

To address these and other healthcare quality issues, state and federal agencies have adopted quality measures assessing the underlying structures, processes, and outcomes of care for accountability, public reporting, and value-based payment programs. Despite the inherent appeal of public reporting and transparency of healthcare quality, there remains minimal evidence that public reporting of healthcare



Quality Measures

Medical Clinical Operations, Oncology Value Creation Team, UnitedHealthcare, Sugarland, TX, USA e-mail: tracy_e_spinks@uhc.com

quality measures improves overall healthcare quality, reduces utilization, or influences healthcare consumer behavior [1– 3]. In view of these observations—and to chart a path forward—it is important to consider the health policy and practice patterns that have contributed to historical issues with healthcare safety, appropriateness, effectiveness, and quality.

This chapter examines the history, current state, and desired future state of health policy for quality in oncologic emergency care. It explores known quality issues and upstream drivers and highlights the important role that quality measures can play in addressing these issues. Additionally, it outlines recommendations for measuring quality in oncologic emergency care and proposes healthcare policy changes and quality measures to drive these changes. Finally, it includes a sample case study describing how ED providers can partner with palliative care providers to reduce ED visits.

History and Current State of Health Policy for Quality in Oncologic Emergency Care

Emergency care lacks formal health policy to support the unique needs of patients with a cancer diagnosis. Existing health policy focuses on providers' duty to treat patients in an emergency along with patient access to emergency medical care. This section describes two key drivers of current health policy for emergency medicine—the no-duty-to-treat principle and the Emergency Medical Treatment and Active Labor Act (EMTALA). The sections that follow explore known issues in oncologic emergency care, contributing factors, and historical efforts to measure the quality of US emergency care.

The No-Duty-to-Treat Principle and the Emergency Medical Treatment and Active Labor Act (EMTALA)

The no-duty-to-treat principle—the controlling law in the USA for over a century—gives physicians significant autonomy in determining which patients they will serve [4]. Several state court cases have supported this principle and have generally held that duty-to-treat begins when the patient-provider relationship is established, regardless of whether the relationship is expressly agreed [5–10]. While the no-duty-to-treat principle remains the controlling law, federal and state entities have established safeguards—through statutes, regulations, and court cases—to prevent discrimination and to ensure access to emergency care [4, 11–18].

Enacted through the Consolidated Omnibus Budget Reconciliation Act of 1986 [19], EMTALA remains the most influential US law affecting emergency care. An "antidumping" law, EMTALA prohibits discriminatory practices in emergency care. The law obligates Medicare-participating hospitals with dedicated EDs to screen, stabilize, and, where necessary, accept transfer patients, regardless of insurance status, ability to pay, or existing relationship with the facility. EMTALA prohibits the transfer of medically unstable patients for any reason except medical necessity (e.g., for specialized emergency care that is unavailable at the transferring facility). EMTALA's screening, stabilization, transfer, and recipient provisions are considered absolute unless modified due to a disaster or public health emergency, such as the 2019 novel coronavirus outbreak [20, 21]. Violations carry heavy fines and can lead to suspension from the Medicare program.

Over time, EMTALA's provisions have been clarified through regulations, court cases, and statutes, including the Patient Protection and Affordable Care Act of 2010 (ACA). These actions clarified hospital responsibilities to staff oncall specialists and defined the physical locations that fall within EMTALA's jurisdiction. Importantly, protections under EMTALA do not apply to outpatients with scheduled nonemergency procedures, and hospitals' stabilization and transfer rights and duties terminate once a patient is admitted as an inpatient [4, 22-31]. Many EMTALA provisions remain controversial. For example, EMTALA facilitates critical access to emergency care but makes no provision for reimbursement of that care. Uncompensated emergency care exceeds uncompensated inpatient and outpatient care combined [32] and remains the burden of emergency care providers [33, 34]. Additionally, EMTALA does not address quality of care. Thus, as long as emergency care is delivered in good faith, misdiagnosis, delays, and medical negligence do not constitute EMTALA violations. EMTALA's stabilization obligations are considered absolute, even when such care is futile due to an underlying condition or when it conflicts with standard of care or a physician's moral and ethical judgment [4, 26, 30].

In summary, the no-duty-to-treat principle and EMTALA represent a strong policy framework to ensure patient access to emergency medical care in the USA. Nearly 35 years after its passage, EMTALA continues to provide essential protections to the nation's most vulnerable residents—the uninsured, underinsured, and disenfranchised—by prohibiting discriminatory emergency care. EMTALA's lasting consequence is the transformation of the US emergency care system into a de facto safety net to address access and financial barriers to primary care. A predictable, albeit unintended, outcome is that EDs are overloaded and inadequately funded to comply with this federal mandate [35]. This compromises the quality and accessibility of emergency care for all patients, including those with a cancer diagnosis. Quality issues for oncologic emergency care are described in the next section of this chapter.

Known Quality Issues

In pursuing emergency care, patients with a cancer diagnosis encounter a mix of quality issues—both oncologic-specific and generalized. Moreover, many of these issues arise from inadequate access to care or quality issues in primary care and outpatient settings. Six issues that affect cancer patients in the emergency setting are described below: (1) late-stage cancers presenting to the ED; (2) overutilization of emergency services; (3) overcrowding, boarding, and diversion; (4) high costs at the end of life; (5) patient dissatisfaction with emergency care; and (6) caregiver burden. Specific issues for dedicated oncologic EDs are also discussed in this section.

Late-Stage Cancers Presenting to the ED

In a well-coordinated healthcare system, patients receive routine primary care and guideline-based cancer screenings, and cancer is diagnosed at an early stage in the primary care setting. However, many undiagnosed cancers present to the ED each year [36-40], with 11% of breast, prostate, colorectal, and lung cancers diagnosed in US EDs from 2004 to 2013 [41]. This is problematic for several reasons. First, these patients often present with generalized symptoms (e.g., nausea and vomiting, fatigue, and bleeding) that may indicate multiple underlying conditions. Moreover, emergency physicians may lack established relationships and a comprehensive medical background for these patients. Therefore, an accurate diagnosis and treatment may be further delayed until the patient seeks follow-up outpatient care or returns to the ED with continued symptom escalation. Second, cancers diagnosed in the ED tend to be of later stage and, therefore, poorer prognosis. Researchers have observed worsened outcomes, including lower overall survival, higher perioperative mortality and readmissions, and longer length of stay [37, 38]. Third, ED-diagnosed cancers suggest disparities in healthcare. For example, researchers at a Florida safety net hospital observed that African Americans, urban dwellers, and those without private insurance were more likely to have a cancer diagnosed in the ED and were at increased risk for stage IV cancer and death [38]. Similarly, a Michigan study of ED-based lung and colorectal cancer diagnoses noted that cancer diagnoses in the ED were disproportionate among older persons, African Americans, dual-eligible patients (patients eligible for Medicare and Medicaid benefits), and

patients with three or more comorbidities. Of note, patients diagnosed with cancer in the ED had significantly more healthcare encounters (inpatient, outpatient, and primary care) in the months preceding their diagnosis [36]. This suggests a gap in the quality, rather than the quantity, of healthcare services. These findings highlight gaps in the nation's population health strategies and indicate opportunities for improvements in screening, early detection, care coordination, and patient education—particularly for more vulnerable populations.

Overutilization of ED Services

Cancer patients present to the ED with acute conditions, including sepsis, spinal cord compression, deep vein thrombosis, and respiratory and gastrointestinal obstruction. Emergency physicians, trained to diagnose and treat acute illness and injury and to stabilize patients for further treatment, are uniquely qualified to manage these emergencies. However, not all emergency visits by oncologic patients represent true oncologic emergencies. Duflos et al. and Wallace et al. observed potentially avoidable emergency visits at 48% and 52% of ED presentations, respectively [42, 43]. Also, Delgado-Guay et al. observed that patients receiving palliative care had a lower rate of avoidable ED visits (23%) [44]. Together, these findings suggest overutilization of emergency services for symptoms associated with progression of disease and treatment side effects that could be managed effectively in the outpatient setting, particularly with palliative care integration. Cancer patients often present with several interrelated symptoms, including pain, fatigue, dyspnea, nausea, dehydration, depression, and cognitive impairment. Chronic pain, in particular, is a frequent complaint among cancer patients visiting the ED. In the absence of highly coordinated multi-symptom management, cancer patients experience frequent ED visits, especially near the end of life [45]. Racial and socioeconomic disparities are a factor here as well [46]. Several observational studies have examined ED service utilization among cancer patients at the end of life. The findings of these studies vary—27–37% of the study populations had an ED visit in the last 14 days of life, and 7-19% of the study populations experienced multiple ED visits in the last 30 days of life [47]. In a 2010 study of hospice enrollees, Carlson et al. observed that patients who disenrolled from hospice were significantly more likely to have an ED visit compared to their continuously enrolled counterparts (33.9% vs. 3.1%) [48].

In the early 2000s, Earle et al. identified frequent ED visits as an indicator of poor quality of care [49, 50]. Aprile et al. concluded that over 50% of unplanned visits at an acute oncologic clinic were repeat presentations [51]. Repeat ED visits suggest healthcare quality issues across care settings. For example, patients may receive inadequate symptom management, discharge instructions, or follow-up care as part of the initial ED visit. Alternatively, upstream access or care coordination issues may lead patients to seek emergencybased care that could be delivered in a less costly outpatient setting. In other cases, repeat ED visits indicate that patients-particularly those with complex comorbidities, diminished performance status, or poor prognosis-are receiving overly aggressive treatment (e.g., chemotherapy), where the treatment toxicities exceed the potential clinical benefits. Repeat ED visits may also indicate delayed access to hospice and palliative care services or that caregivers are not prepared to manage and cope with the burden of cancer disease at home. These trends highlight the need for more selective use of aggressive treatment, improved symptom management, and earlier introduction of advance care planning. Likewise, greater access to palliative and hospice care, same-day/next-day physician appointments, and 24/7 access to providers may reduce ED utilization by cancer patients, particularly at the end of life. These care delivery approaches are discussed later in this chapter.

Overcrowding, Boarding, and Ambulance Diversion

The demand for emergency services routinely exceeds ED capacity, particularly in large urban areas. Due to coalescing system-level issues, including ED closures, inadequate or delayed access to primary and specialty care, and higher rates of uninsurance and underinsurance, ED overcrowding has worsened over time [52]. ED crowding is worsened by ED "boarding," where admitted patients remain in the ED for hours-even days-until a hospital bed becomes available. Sadly, ED boarding has become a routine practice for most EDs and reflects high inpatient census rates and inefficient admission processes [35]. More than an inconvenience, ED overcrowding and extended ED boarding have been associated with poorer quality of care and patient experience and may disproportionately affect older, sicker patients and those requiring specialized inpatient care [53]. Key indicators include treatment delays, patients leaving the ED without being seen, increased risk for medical errors (including medication errors), and multiple poorer outcomes, including longer lengths of stay and higher inpatient mortality rates [53-57].

Unmanaged ED crowding and prolonged ED boarding contribute to ambulance diversion. Once a practice reserved for catastrophic events, diversion has become increasingly common, especially in urban areas. By delaying treatment or by redirecting patients to EDs without the resources and expertise to optimally care for their severity of illness, diversion can place patients with acute conditions at significant risk [35]. Furthermore, extended diversion time has been associated with adverse patient outcomes, particularly for patients with life-threatening conditions [58, 59]. Together, ED overcrowding, extended boarding, and ambulance diversion increase stress among ED providers along with patients' risk for adverse events and poorer outcomes. Accordingly, experts have advocated for stronger standards to reduce these practices [35]. While these findings and recommendations are not specific to oncologic emergency care, they have important implications for cancer patients seeking ED care.

High Costs at the End of Life

In 2010, an estimated \$38 billion was spent on end-of-life care for cancer patients in the USA. By 2020, those costs were projected to increase to between \$49 billion and \$74 billion, representing up to 36% of total spending for cancer care [60]. Underlying factors include healthcare fragmentation, frequent transitions between care settings, inadequate care coordination, no or delayed access to palliative and hospice care, and overutilization of aggressive treatment for patients with advanced disease. Additionally, despite efforts to transition to value-based care, most cancer care is still reimbursed as fee-for-service, which creates financial incentives for providers to deliver high-cost and high-intensity services, even at the end of life. Vera-Llonch et al. estimated total healthcare spending at nearly \$126,000 and \$129,000 for patients receiving chemotherapy for metastatic lung cancer and metastatic breast cancer, respectively [61, 62]. A study of patients with stage IV breast, colon, lung, and prostate cancers observed that one-third of patients received a high-cost advanced imaging study (computerized topography or CT, magnetic resonance imaging or MRI, positron emission tomography or PET, and nuclear medicine or NM) in the last month of life, with the top 10% receiving three of these imaging studies [63]. A multinational study on site of death found that the USA had the lowest proportion of inpatient deaths among cancer patients (22%) but double the ICU admissions and length of stay in the last 30 and 180 days of life [64]. These findings suggest a higher intensity of care at the end of life in the USA when compared with other developed nations [65]. A recent study of patients with metastatic non-small cell lung cancer in the National Cancer Database indicated that 10.8% and 24.5% of patients initiated first-line treatment within 4 and 8 weeks of death, respectively [66]. Aggressive treatment at the end of life contributes to unsustainable national healthcare spending on end-of-life care without comparable benefit in terms of better survival, quality of life, and access to care. Since Americans have ranked treatment costs and financial burden to family members as their biggest concerns when faced with a life-limiting illness [67, 68], excessive healthcare costs exacerbate emotional distress among patients with a poor prognosis.

Significant geographic and between-hospital variation in end-of-life costs has been observed. A seminal study by the Dartmouth Atlas Project identified the availability of healthcare resources, rather than patient acuity or patient preference, as the most significant contributing factor [69]. A committee convened by the National Academy of Medicine, or NAM (neé Institute of Medicine), attributed 89% of variation in total Medicare spending to variation in acute care and post-acute care [70]. These findings have important implications to begin addressing overutilization of services (including ED visits) at the end of life and suggest that better care coordination may reduce spending for these patients.

Patient Dissatisfaction with Emergency Care

ED overcrowding, poor patient handoffs, and extended wait times-perceived and actual-compromise patient experience and contribute to patients leaving the ED without being seen [45, 71, 72]. Patient experience with ED care has not been systematically measured in the USA. However, researchers within and outside the USA have identified promoters and detractors of patient satisfaction with emergency care, with mixed results [73-75]. Overall satisfaction has been associated with provider interpersonal skills-communication, courtesy, empathy, and competence-and patient perception regarding wait times [75-78]. ED physicians often lack an established relationship with patients and carry heavy patient loads of varying acuity, leading to significant challenges to timely and accurate communication [79]. Patient satisfaction may be improved by training ED providers to initiate more frequent and targeted communication, particularly regarding wait times, and by expanding ED provider access to patient records across care delivery systems.

Some studies have demonstrated higher satisfaction among ED patients of higher acuity (and vice versa) [80–82]. Additionally, lower-acuity patients have expressed greater dissatisfaction with wait times and costs of care than their higher-acuity counterparts [82]. This difference may be attributed to two factors. First, urgent or emergent ED patients likely will be triaged more quickly than their nonurgent counterparts. Second, the fact that lower-acuity patients could be seen more quickly—and at a lower cost—in an outpatient setting may contribute to their dissatisfaction. Redirecting lower-acuity patients from the ED to more appropriate outpatient settings may help address this issue along with easing ED overcrowding and reducing total costs of care.

In 2012, the Centers for Medicare & Medicaid Services (CMS) funded initial work on a Consumer Assessment of Healthcare Providers and Systems (CAHPS) survey for 41

emergency care-the ED CAHPS (formerly Emergency Department Patient Experience of Care, or EDPEC, Survey). Development, testing, and refinement of the survey were completed in 2020. The 35-item survey, which assesses frequency of positive experience with ED processes, staff, healthcare received, and overall experience, received the CAHPS® trademark in March 2020. The ED CAHPS, Version 1.0, includes four composite measures (getting timely care, how well doctors and nurses communicate, communication about medications, and communication about follow-up) and two global measures (overall ED rating and willingness to recommend the ED) for patients discharged to home [83]. Currently, ED CAHPS use is voluntary, and providers are not required to report their results publicly. Nevertheless, early adoption of this survey will yield important insights regarding opportunities to improve patient experience with ED care.

Caregiver Burden

When caring for relatives with debilitating and chronic conditions, such as cancer, family caregivers experience significant financial, social, physical, and psychological distress. As cancer care continues to shift to outpatient settings, caregivers face increasing pressure to help their loved one navigate a complex and fragmented care delivery system and to care for their loved one at home while receiving limited training and support [84]. A 2011 survey from AARP, Inc. and the United Hospital Fund noted that 46% of caregivers of patients with multiple chronic conditions reported performing medical care (e.g., medication management and operating specialized medical equipment) for their loved one. Additionally, 53% of caregivers reported serving as care coordinators [85].

Several studies have described morbidity in caregivers of cancer patients [86]. Braun et al. reported that nearly 39% of caregivers of patients with advanced cancer experienced significant symptoms of depression [87]. Grunfeld et al. observed that caregivers of patients with advanced breast cancer experienced anxiety and depression that matched or exceeded the patient's anxiety and depression [88]. Wright et al. noted that place of death affected caregiver well-being. Death in the ICU or inpatient setting increased caregiver risk for post-traumatic stress and prolonged grief when compared with death at home [89]. Researchers have also described lifestyle interference among caregivers of cancer patients. Wadhwa et al. determined that 25% of caregivers of persons with advanced cancer experienced a change in work status [90], while Mazanec et al. estimated a 23% loss of work productivity among caregivers [91]. This is problematic, since increased lifestyle interference increases caregiver emotional distress [92] and can compromise their ability to provide logistical and emotional support to the cancer patient [93]. Caregiver emotional distress can also negatively affect patient well-being. Two longitudinal studies of partners of breast cancer patients revealed that patient fatigue, symptom distress, anxiety, and depression increased in parallel with caregiver emotional distress [94, 95].

To prepare family members for caregiver role demands, the NAM recommended that healthcare agencies, including the Department of Health and Human Services (HHS), fund demonstration projects to train caregivers of cancer patients for their demanding role [84]. Federal and state lawmakers have also prioritized caregiver support through recent legislation-the Recognize, Assist, Include, Support, and Engage (RAISE) Family Caregiver Act of 2017 [96] and Caregiver Advise, Record, Enable (CARE) Act, signed into law in 39 states [97]. Yet, additional research and real-world implementation guidance are needed for meaningful change. Experts have identified interventions focused on caregiver skills and decision support, stress reduction, interdisciplinary palliative care, and psychosocial support to reduce caregiver burden, but such interventions remain largely untested [98, 99]. Caregiver training should focus on evolving caregiver needs throughout the continuum of care since shortterm caregiver training has shown promising results [100]. but caregiver-perceived preparedness, quality of life, and psychological well-being may diminish over time [101]. Professional and patient advocacy organizations can play an important role in developing educational materials and supporting programs to help caregivers manage their distress and provide optimal support for cancer patients [98].

Specific Issues for Dedicated Oncologic EDs

Dedicated oncologic EDs face additional pressures related to access and care coordination. Some patients with a cancer diagnosis seek entry to a comprehensive cancer center through the center's dedicated ED, if one exists [102, 103]. Additionally, EDs at other hospitals may seek to transfer uninsured or underinsured cancer patients to a specialized cancer center through its dedicated ED on the basis of an oncologic emergency that the transferring center is unable to manage. In these cases, the ED may serve as an interface or gateway into specialized oncologic care systems. However, there is no guaranteed access to oncologic care. Under EMTALA, the dedicated oncologic ED has the duty to screen and stabilize the patient in the ED. However, there is no duty to admit the patient, once stabilized, for further treatment of the patient's cancer diagnosis or other health issues. Even if active cancer treatment (e.g., chemotherapy) is required to stabilize the patient, there is no requirement to continue the treatment beyond the initial emergency presentation [104]. Thus, cancer patients may be bounced between multiple care settings, placing them at greater risk for receiving unsafe and poorly coordinated care.

Upstream Drivers

In the preceding section of this chapter, we discussed six quality issues affecting oncologic emergency care and described specific issues for dedicated oncologic EDs. Often, these issues arise when cancer patients seek ED care, but they are more directly associated with inadequate access to care or primary care gaps. Six upstream drivers that compromise ED-based oncologic care are described below: (1) poor care coordination; (2) underutilized advance care planning; (3) inadequate access to palliative care; (4) delayed hospice referral and the hospice reimbursement model; (5) limited availability of immediate and after-hours outpatient care; and (6) unrealistic patient/caregiver expectations regarding prognosis and treatment.

Poor Care Coordination

Fragmented healthcare delivery and inadequate care coordination are common among the elderly, the uninsured and underinsured, and patients with chronic and life-threatening conditions. Cancer patients may experience fragmented care across the cancer care continuum [105–109] as they frequently move between oncologic care, primary care, community and specialty hospitals, EDs, hospice, and long-term care. Indeed, the NAM identified nearly 30 clinical roles and disciplines involved in cancer care [84]. Specific issues relate to follow up on abnormal findings, management of comorbid conditions and symptom burden, medication administration (including chemotherapy), psychosocial support, end-of-life care, and survivorship transitions [105–116]. Increasingly, hospital outpatient departments are the care setting for complex cancer treatment [117], including stem cell transplantation, cancer surgery, and immunotherapy. Shifting these services from inpatient to outpatient settings has many benefits for patients, reduces healthcare costs, and eases the demand for inpatient resources. However, it places patients at increased risk for unmanaged pain, infection, febrile neutropenia, anemia, dehydration, nausea and vomiting, gastrointestinal distress, and dyspnea that lead patients to seek care in the ED. Thus, ED visits and, in particular, repeat ED visits may indicate unmet patient needs in other settings or that caregivers are unprepared to care for their loved one's disease at home. This is especially true at the end of life, when cancer patients present at the ED with poorly managed symptoms or progression of disease.

In the face of inadequate care coordination by the primary oncologic team, ED providers may become de facto oncologic care coordinators. However, ED physicians are trained to manage acute injury and illness and to stabilize patients for further treatment. Moreover, emergency physicians may be uncomfortable with addressing end-of-life issues in cancer patients [71]. Therefore, overextended physicians feel increasing pressure to ensure that patients are directed to appropriate follow-up care (including hospice or palliative care) and to close the loop with primary care physicians and oncologic providers. Strong care coordination, starting with a well-defined interface between primary care providers and oncologists [108, 109], is imperative to address fragmented cancer care. Both formal and informal primary care/oncologic relationships can improve care coordination [118], leading to improved handoffs and more appropriate healthcare utilization [119] and lessening the burden on ED care teams.

Underutilized Advance Care Planning

Advance care planning enables patients to consider their end-of-life preferences; to communicate those preferences to their family members, caregivers, and healthcare providers; and to document their preferences regarding life-sustaining procedures in a legally binding advance directive. Ideally, advance care planning begins during treatment planning, is documented by providers using structured templates [120], is informed by educational aids [121], and is revisited throughout treatment at clinically relevant time points (e.g., if the patient's prognosis worsens). In the context of cancer treatment, it should include ongoing communication between patients, caregivers, and providers across care delivery settings. The National Comprehensive Cancer Network (NCCN) recommends early initiation of advance care planning using a tailored approach based on where the patient is in his/her cancer trajectory [122]. For patients with advanced disease, advance care planning is critical to delivering patient-centered care and is essential to align treatment plans with patient values and preferences for quality of life, treatment intensity, and life-prolonging treatment. Early findings indicate that advance care planning has several benefits: reduced aggressive treatment and increased hospice referral at the end of life [123]; fewer ICU admissions; hospitalizations and inpatient deaths in the last 30 days of life [124]; clearer and earlier understanding of patient prognosis [125]; better alignment between patient preferences and care at the end of life [126]; and improved satisfaction and reduced stress and anxiety for patients and their families [127]. Despite fears to the contrary, it is not associated with increased hopelessness and anxiety for patients with advanced disease [128, 129].

Despite the potential benefits of advance care planning, end-of-life care discussions are often delayed until all curative treatment options are exhausted [130] and death is imminent [131]. Furthermore, researchers have observed large proportions of cancer patients presenting to the ED without an advance directive [132, 133]. Even when an advance directive exists, ED providers may be unable to access it in time to honor the patient's wishes regarding life-prolonging treatment when the patient presents with a sudden acute, life-threatening complication or critical decline in health status.

Of note, efforts to improve advance care planning have focused on executing advance directives for patients with poor prognosis. Advance directives are integral to advance care planning. However, advance care planning is much broader [122, 134] and includes thoughtful consideration of patient preferences regarding life-sustaining procedures, treatment intensity, quality of life at the end of life, and place of death. Goswami et al. and Pajka et al. have shown promising results in engaging advance practice nurses and ED staff in advance care planning [135, 136]. Future efforts should focus on coordinated, systematic, and patient-centered approaches to initiate advance care planning much earlier in the trajectory of disease, especially for patients with laterstage diagnoses.

Inadequate Access to Palliative Care

Palliative care addresses the physical and psychosocial effects of the disease and its treatment, thereby easing the burden of cancer throughout the continuum of care. Researchers have proposed early introduction of palliative care as an important strategy to improve symptom management and quality of life [137, 138], to help patients have more realistic expectations regarding their cancer [139], to reduce healthcare spending and utilization of acute care and emergency services [140, 141], and to improve survival in some patients [142]. Conversely, poor health-related quality of life has been associated with worse survival [143-146]. In the twenty-first century, the USA has seen substantial growth in palliative care programs [147-150]. Between 2009 and 2018, outpatient palliative care programs increased among NCI-designated cancer centers (59-95%) and non NCIdesignated cancer centers (22-40%) [149]. Similarly, in 2019, inpatient palliative care programs were identified at 72% of hospitals with 50 or more beds (vs 67% in 2015 and 7% in 2001). This growth was concentrated among hospitals that are large (300 or more beds), urban, public or nonprofit, and located in northeastern states [150]. Thus, despite this increased capacity, palliative care services are not readily accessible for many cancer patients or may not be offered on a timely basis. Experts have suggested that palliative care referrals may be delayed due to erroneous perceptions among oncologists that palliative care and curative treatment must follow sequential pathways. Consequently, palliative care needs remain unmet, and patients with distress associated

with advanced disease or high symptom burden frequently seek care in the ED, particularly at the end of life.

To reduce barriers to timely palliative care, researchers have recommended colocation of outpatient oncologic and palliative care [151, 152] along with integrating palliative care with ED services [153–155]. Early studies suggest the benefits of ED-based palliative care programs, including better quality of life, reduced length of stay, lower intensive care, and improved hospice utilization, though the data are conflicting [156–159]. Researchers have identified several barriers to ED-integrated palliative care; these include inadequate palliative care training, an ED culture that favors aggressive treatment, and provider fear of being sued [160– 163]. A multi-stakeholder workgroup convened in 2009 by the Agency for Healthcare Research and Quality (AHRQ) and the American College of Emergency Physicians (ACEP) identified four research priorities to address these barriers:

- 1. Which patients are in greatest need of palliative care services in the ED?
- 2. What is the optimal role of emergency clinicians in caring for patients along a chronic trajectory of illness?
- 3. How does the integration and initiation of palliative care training and services in the ED setting affect healthcare utilization?
- 4. What are the educational priorities for emergency clinical providers in the domain of palliative care? [164]

Interest in these areas has surged in recent years, particularly in testing different models of ED-based palliative care and the impact on healthcare utilization. Perhaps the greatest progress has been in the development and dissemination of guidelines, toolkits, and educational materials to advance palliative care in the ED [163, 165–171]. Continued experimentation and socialization are needed to drive clinical transformation in this important field.

Delayed Hospice Referral and the Hospice Reimbursement Model

Hospice programs that offer team-based comprehensive and interdisciplinary care can improve comfort and quality of life for cancer patients with a life expectancy of 6 months or less. Moreover, hospice referral is associated with higher quality care, including fewer hospitalizations, ICU admissions, and inpatient deaths and lower costs of care [172, 173]. Electing hospice care requires Medicare beneficiaries to forgo curative treatment and is appropriate when the risks or complications of treatment outweigh the potential benefits. Hospice referrals have increased significantly since the Medicare hospice benefit was created by the Tax Equity and Fiscal Responsibility Act of 1982 [174]. Over 1.5 million Medicare beneficiaries, representing over 50% of Medicare decedents, received hospice services in 2018. Cancer patients continued to lag behind their non-cancer counterparts in terms of hospice utilization. The share of hospice decedents with cancer declined from 52% to 26% between 2000 and 2018. In 2018, average and median length of hospice stay for cancer decedents were 53 and 17 days, respectively, versus average and median length of stay for decedents with neurological conditions (151 and 38 days, respectively). Together, these factors indicate that many cancer patients are enrolling in hospice too late to benefit fully from the team-based comprehensive and interdisciplinary palliative care that hospice programs offer [175].

Several barriers have been identified to earlier hospice referral. These include financial incentives to keep patients in the acute care system, provider discomfort with initiating end-of-life discussions, and patient and family difficulty accepting a terminal cancer prognosis [175]. Desired intensity of care remains a significant barrier to earlier hospice enrollment in the USA where patients with a terminal disease (defined as a life expectancy of 6 months or less) must agree to forgo curative treatment to qualify for the Medicare hospice benefit. Once patients are enrolled, Medicare pays hospice providers a per diem rate per enrollee. Under this capitated model, reformed in 2016 to better align reimbursement with the intensity of care delivered [176], hospice providers assume financial responsibility for all care related to the patient's terminal illness. Patients with advanced cancer often benefit from palliative radiation and chemotherapy, opioids, and parenteral nutrition. The treatment costs may be substantial and greatly exceed the Medicare hospice benefit. Accordingly, hospice providers may be discouraged from enrolling high-cost cancer patients, or they may implement restrictive enrollment policies aimed at cost control. These restrictions present many patients and caregivers with the dilemma of electing hospice care or comfort care at the end of life [177].

Concurrent cancer treatment and hospice care has been proposed as an effective remedy to address the limitations of hospice benefit design. In the early 2000s, the Robert Wood Johnson Foundation funded 22 demonstration projects of concurrent curative treatment and palliative care. These projects demonstrated early feasibility of an integrative approach to cancer treatment and hospice care across various patient populations and care settings [178]. Aetna conducted a similar pilot-extending hospice eligibility to patients with a life expectancy of 12 months or less-and observed increased hospice enrollment, lower utilization of acute care services, and a 22% reduction in costs [179]. Through its Comprehensive End-of-Life Care Initiative, started in 2009, the Veterans Health Administration offers concurrent cancer treatment (chemotherapy or radiation) and hospice services with good results. Studies conducted in patients with metastatic non-small cell lung cancer demonstrated increased hospice utilization, less aggressive treatment at the end of life, and lower costs of care [180-182]. Success was associated with preserved hope and better quality of life among patients along with smoother transitions, stronger care coordination, and more frequent touch points with care teams. Concern regarding compliance with Medicare's either-or policy (i.e., cancer treatment or hospice) was cited as a potential barrier [183]. In 2016, the Center for Medicare and Medicaid Innovation (CMMI) launched the 4-year Medicare Care Choices Model to test concurrent cancer treatment and hospice in the Medicare and dual-eligible population (as authorized under the ACA) [184]. Experts have identified design limitations, including restrictive eligibility criteria. Early findings have revealed both challenges and successes. Hospice attrition initially was high, and beneficiary enrollment is much lower than expected. Advance care planning has been widely utilized. Both patients and caregivers have reported positive experience with care. Perhaps the pilot's greatest success has been improved utilization of hospice services (including counseling and symptom management) among enrolled cancer decedents. This suggests that concurrent cancer treatment and hospice may serve as a bridge to traditional hospice enrollment by offering patients and caregivers more time to process the patient's prognosis and treatment options while receiving supportive care [185, 186]. Additional demonstration projects are needed across federal, state, and commercial insurers to ensure that hospice benefit design promotes better quality of life, timely hospice enrollment, and, where appropriate, concurrent cancer treatment and hospice.

Limited Availability of Immediate and After-Hours Outpatient Care

Experts suggest that many ED visits are for non-emergent complaints that are more appropriately managed in the outpatient setting. Hansagi et al. observed that two-thirds of ED patients in their study were primary care cases, but the patients could not get in to see their physician or were referred to the ED for care [81]. Similarly, in an observational study of ED visits in North Carolina, Mayer et al. found that 44.9% of ED visits occurred during normal clinic hours and fewer than one-fifth of those patients were admitted to the hospital [187]. These findings underscore the need for more immediate access to outpatient oncologic care, such as through same-day/next-day appointments or 24/7 provider access.

The effectiveness of these practices is being tested through oncologic-specific patient-centered medical homes (PCMH). A primary care delivery model, the PCMH is designed to provide comprehensive, well-coordinated, patient-centered care by promoting access to preventive, chronic, and acute care, as well as a systems-based approach to safety and quality [188]. The Patient-Centered Outcomes Research Institute funded a 5-year PCMH model pilot with five medical oncologic practices in Pennsylvania. The pilot demonstrated greater access to specialty care and improved patient experience with shared decision-making but no change in ED visits and inpatient admissions [189, 190]. A broader pilot-Community Oncologic Medical HOME (COME HOME)funded by CMMI [191] piloted similar approaches, including active disease management, triage and clinical pathways, enhanced provider access, and interdisciplinary teams, to deliver more coordinated cancer care. This pilot, implemented in seven community oncologic practices, produced a 10.2% reduction in ED visits and lower costs of care [192, 193]. These models of care should be studied further to determine the generalizability of these approaches to cancer care in the community and at academic medical centers.

Unrealistic Patient/Caregiver Expectations Regarding Prognosis and Treatment

Patient preference for intensity of treatment is influenced by health literacy, provider mistrust, family dynamics, religious beliefs, and other cultural and religious factors [194]. Cancer patients must have an accurate understanding of their treatment options and prognosis to avoid unnecessary, futile care and make treatment decisions that are consistent with their preferences and values. Indeed, patients that overestimate their prognosis are more likely to receive aggressive treatment of questionable benefit [195], while patients who understand their prognosis prefer symptom-directed care [131]. Several studies have confirmed that patients with advanced disease frequently misunderstand the intent of their cancer treatment and overestimate their prognosis [137, 196-198]. For example, Temel et al. published a study of newly diagnosed patients with metastatic lung cancer in 2011, noting that 32% of respondents considered their cancer curable and that 69% of respondents believed they were receiving curative, rather than palliative, treatment [139]. Likewise, Weeks et al. reported that 69% and 81% of patients with metastatic lung and colorectal cancer, respectively, did not understand that they were receiving palliative chemotherapy [195].

Communication challenges between patients, their caregivers, and providers contribute to patient and caregiver misunderstandings about prognosis or treatment intent. In some cases, patients receive accurate prognostic information, but do not understand or do not accept their prognosis. In other cases, physicians may be reluctant to provide this information, will do so only when asked by the patient, or will provide inflated survival estimates to their patients [195, 198, 199]. Mack and Smith attributed provider communication issues to discomfort with these discussions, uncertainty in estimating prognosis, and concerns regarding reduced hope, patient depression, and cultural appropriateness [200]. In 2013, the NAM recommended five strategies for improving patient-centered communication and shared decision-making for cancer patients:

- 1. Making more comprehensive and understandable information available to patients and their families
- 2. Developing decision aids to facilitate patient-centered communication and shared decision-making
- 3. Prioritizing clinician training in communication
- 4. Preparing cancer care plans
- 5. Using new models of payment to incentivize patientcentered communication and shared decision-making [84]

Implementing these approaches will assist providers in communicating more clearly regarding treatment intent and prognosis and will contribute to more realistic assessments among patients and their caregivers. Moreover, physicians should seek to understand their patients' preferences for prognostic information and adapt their communication styles accordingly.

Role of Quality Measures

Healthcare quality measures offer quantitative and qualitative assessments of the healthcare delivery. Experts have developed healthcare quality measures to evaluate the underlying structures and processes, outcomes, patient experience, and, to a limited degree, the costs of care. Moreover, there is increasing interest in measuring caregiver burden and experience with care. Some measures are developed for specific health conditions (e.g., breast cancer) or care delivery settings (e.g., ED), while other measures are crosscutting and apply to multiple health conditions or care delivery settings.

In this section, we describe the history of quality measurement in emergency medicine, provide examples of existing ED quality measures that are relevant to cancer care, and discuss the limitations of these measures.

History of Quality Measurement in Emergency Medicine

National quality measurement for emergency medicine began in the early 2000s as part of CMS' Reporting Hospital Quality Data for Annual Payment Update (RHQDAPU) program. The RHQDAPU program was a voluntary CMS quality reporting program that became the Inpatient Quality Reporting (IQR) program in 2010. The Medicare Prescription Drug, Improvement, and Modernization Act of 2003 (MMA) introduced financial incentives for hospitals to report data on ten quality measures for pneumonia, acute myocardial infarction (AMI), and congestive heart failure via the RHODAPU program [201, 202]. These measures were developed through the Hospital Quality Alliance, a public/ private partnership with membership from CMS, the The Joint Commission, the American Hospital Association, and healthcare consumer groups [203]. In 2004, these data were published as the first national comparative dataset for ED quality. The financial incentives created under the MMA were later strengthened by the Deficit Reduction Act of 2005 (DRA) [204] and expanded to include measures for hospitalbased outpatient care under the Tax Relief and Health Care Act of 2006 [205]. In 2015, Congress consolidated physicianfocused federal quality programs through the Merit-based Incentive Payment System (MIPS), a key component of the Medicare Access and CHIP Reauthorization Act of 2015 (MACRA) [206].

Over time, ED quality measurement has been expanded through public and private sector efforts. In 2006, the American Medical Association's Physician Consortium for Performance Improvement (AMA-PCPI), ACEP, and NCQA co-developed physician-level ED measures for pneumonia. chest pain, and syncope [207, 208]. Additional independent measure development projects were undertaken by hospitals, by CMS, and by professional organizations, such as ACEP. These efforts focused on specific aspects of care (e.g., timeliness of care and ED communication). Disease-specific measures of morbidity, mortality, and resource use were also developed [208, 209]. At two Performance Measures and Benchmarking Summits convened in 2006 and 2010, participants proposed a wide range of metrics: operational metrics (e.g., ED census), timestamp and interval metrics (e.g., ED length of stay), proportional metrics (e.g., left without being seen), and utilization metrics (e.g., specialty consultations) [210, 211]. Stone-Griffith et al. developed the ED Dashboard and Reporting Application to support data-driven ED performance improvement projects by routinely measuring ED throughput [212].

In parallel, the National Quality Forum (NQF), a nonprofit organization that uses a consensus development process to endorse healthcare quality measures for use in federal public reporting programs, launched a two-phase project endorsing a national measure set for ED care. Between 2007 and 2009, the NQF endorsed 22 measures for ED care, including nine measures that were given time-limited (or temporary) endorsement and pending completion of measure testing and validation [213]. Subsequent projects focused on regionalized emergency care, care transitions, and chief complaint-based performance assessments [214–217]. These measures are included in Table 4.1. Some of these measures have been adopted for CMS public reporting programs,

	Relevant to cancer	Yes	°Z	°Z	No	°Z
	CMS *	Program PQRS, QHP QRS		PQRS, QRUR, BPM	PQRS, QRUR, BPM	
		Endorsed	Endorsement removed (October 2012)	Endorsement removed (<i>April</i> 2017)	Endorsement removed (December 2017)	Endorsement removed (<i>February</i> 2016)
	Year	2009	2009	2009	2007	2007
	Measure target	(process) (process)	Asthma (process)	Musculoskeletal (process)	Cardiovascular (process)	Cardiovascular (process)
	Measure	NCQA	IPRO	NCQA	AMA-PCPI	AMA-PCPI
-	- - -	Health plan	Clinician	Health plan, integrated delivery system	Group	Group, clinician
•	* • •	The percentage of adolescent and adult members with a new episode of AOD abuse or dependence who initiate treatment through an inpatient AOD admission, outpatient visit, intensive outpatient encounter, partial hospitalization, telehealth, or medication-assisted treatment (MAT) within 14 days of the diagnosis, the percentage of adolescent and adult members with a new episode of AOD abuse or dependence who initiated treatment and who had two or more additional AOD services or MAT within 34 days of the initiation visit	Percentage of patients for whom there is documentation that a written asthma management plan was provided either to the patient or the patient's caregiver OR, at a minimum, specific written instructions on under what conditions the patient's doctor should be contacted or the patient should go to the emergency room	The percentage of patients with a primary diagnosis of low back pain who did not have an imaging study (plain X-ray, MRI, CT scan) within 28 days of diagnosis	Percentage of patients aged 40 years and older with an emergency department discharge diagnosis of non-traumatic chest pain who had a 12-lead electrocardiogram (ECG) performed	Percentage of patients, regardless of age, with an emergency department discharge diagnosis of acute myocardial infarction (AMI) who had documentation of receiving aspirin within 24 h before emergency department arrival or during emergency department stay
`	*	Initiation and Engagement of Alcohol and Other Drug Abuse or Dependence Treatment	Management Plan for People with Asthma	Use of Imaging Studies for Low Back Pain	Emergency Medicine: 12-Lead Electrocardiogram (ECG) Performed for Non-traumatic Chest Pain	Emergency Medicine: Aspirin at Arrival for Acute Myocardial Infarction (AMI)
	NQF	6000	0025	0052	0600	0092

(continued)

(continued)
4.1
e
Ta

unt Ver	2									
Releva to canc	care? [§]	No	No	No	No	No	No	Yes	Yes	No
CMS	program [‡]				PQRS, QRUR, BPM			HACRP, HHQRP		MSSP
	Current status	Endorsement removed (<i>February</i> 2014)	Endorsement removed (December 2011)	Endorsement removed (December 2011)	Endorsed	Endorsement removed (October 2012)	Endorsement removed (October 2012)	Endorsed	Endorsement removed (October 2016)	Endorsed
Vear	endorsed	2007	2007	2007	2009	2007	2009	2009	2009	2007
Measure taroet	(type)*	Cardiovascular (process)	Pneumonia (process)	Pneumonia (process)	Behavioral health (process)	Pneumonia (process)	Pneumonia (process)	Care coordination (outcome)	Cancer-specific (outcome)	Cardiovascular (process)
Megcure	owner*.†	AMA-PCPI	AMA-PCPI	AMA-PCPI	AMA-PCPI	CMS	CMS	CMS	ASCO	AHRQ
	Reporting level	Clinician	Clinician	Clinician	Group, clinician	Facility	Facility	Facility	Group	Health plan, integrated delivery system
	Measure description [*]	Percentage of patients aged 60 years and older with an emergency department discharge diagnosis of syncope who had an ECG performed	Percentage of patients aged 18 years and older with the diagnosis of community-acquired bacterial pneumonia with oxygen saturation assessed	Percentage of patients aged 18 years and older with the diagnosis of community-acquired bacterial pneumonia with mental status assessed	Percentage of patients aged 18 years and older with a diagnosis of major depressive disorder (MDD) with a suicide risk assessment completed during the visit in which a new diagnosis or recurrent episode was identified	Percentage of pneumonia patients 18 years of age and older who have had blood cultures performed in the emergency department prior to initial antibiotic received in the hospital	Percentage of pneurnonia patients 18 years of age and older who receive their first dose of antibiotics within 6 h after arrival at the hospital	Percentage of home health stays in which patients used the emergency department but were not admitted to the hospital during the 60 days following the start of the home health stay	Proportion of patients who died from cancer with more than one emergency department visit in the last 30 days of life	Admissions with a principal diagnosis of heart failure per 100,000 population, ages 18 vears and older
	Measure title*	Emergency Medicine: 12-Lead Electrocardiogram (ECG) Performed for Syncope	Assessment of Oxygen Saturation for Community- Acquired Bacterial Pneumonia	Assessment of Mental Status for Community-Acquired Bacterial Pneumonia	Adult Major Depressive Disorder (MDD): Suicide Risk Assessment	Blood Cultures Performed in the Emergency Department Prior to Initial Antibiotic Received in Hospital	Initial Antibiotic Received Within 6 H of Hospital Arrival	Emergency Department Use Without Hospitalization During the First 60 Days of Home Health	Proportion of Patients Who Died from Cancer with More Than One Emergency Department Visit in the Last 30 Days of Life	Congestive Heart Failure Rate (PQI 08)
NOF	Ξ E	0093	0094	0095	0104e	0148	0151	0173	0211	0277

No	Ŷ	oZ	No	No	Yes	(continued)
		Hospital Compare, OQR		Hospital Compare, OQR		
Endorsement removed (August 2014)	Endorsement removed (January 2012)	Endorsement removed (November 2016)	Endorsement removed (August 2014)	Endorsed	Endorsed	
2007	2007	2007	2007	2007	2007	
Cardiovascular (process)	(process)	(process)	Cardiovascular (efficiency)	Cardiovascular (process)	(process)	
CMS	CMS	CMS	CMS	CMS	UMRHRC	
Facility	Facility	Facility	Facility	Facility	Facility	
Percentage of emergency department acute myocardial infarction (AMI) patients or chest pain patients (with probable cardiac chest pain) without aspirin contraindications who received aspirin within 24 h before ED arrival or prior to transfer	Median time from emergency department arrival to administration of fibrinolytic therapy in ED patients with ST-segment elevation or left bundle branch block (LBBB) on the electrocardiogram (ECG) performed closest to ED arrival and prior to transfer	This measure calculates the percentage of emergency department (ED) acute myocardial infarction (AMI) patients with ST-segment elevation on the electrocardiogram (ECG) closest to arrival time receiving fibrinolytic therapy during the ED stay and having a time from ED arrival to fibrinolysis of 30 min or less	Median time from emergency department arrival to ECG (performed in the ED prior to transfer) for acute myocardial infarction (AMI) or chest pain patients (with probable cardiac chest pain)	Median time from emergency department arrival to time of transfer to another facility for acute coronary intervention	Percentage of patients transferred to another healthcare facility whose medical record documentation indicated that required information was communicated to the receiving facility prior to departure (subsection 1) or within 60 min of transfer (subsections 2–7)	
Aspirin at Arrival	Median Time to Fibrinolysis	Fibrinolytic Therapy Received Within 30 Min of ED Arrival	Median Time to ECG	Median Time to Transfer to Another Facility for Acute Coronary Intervention	Emergency Transfer Communication Measure	
0286	0287	0288	0289	0290	0291	

(continued)
4.1
Ø
ab

Relevant to cancer care? [§]	Yes	Yes	Yes	Yes	Yes	Yes	Yes
CMS program [‡]							
Current status	Endorsement removed (<i>September</i> 2014)	Endorsement removed (September 2014)	Endorsement removed (September 2014)	Endorsement removed (September 2014)	Endorsement removed (September 2014)	Endorsement removed (September 2014)	Endorsement removed (<i>April</i> 2014)
Year endorsed	2007	2007	2007	2007	2007	2007	2008
Measure target (type)*	(process)	Care coordination (process)	Care coordination (process)	Care coordination (process)	Care coordination process)	Care coordination (process)	Care coordination (structure)
Measure owner*⁺†	UMRHRC	UMRHRC	UMRHRC	UMRHRC	UMRHRC	UMRHRC	CMS
Reporting level	Facility	Facility	Facility	Facility	Facility	Facility	Facility
Measure description [*]	Percentage of patients transferred to another healthcare facility whose medical record documentation indicated that the entire vital signs record was communicated to the receiving facility within 60 min of departure	Percentage of patients transferred to another healthcare facility whose medical record documentation indicated that medication information was communicated to the receiving facility within 60 min of departure	Percentage of patients transferred to another healthcare facility whose medical record documentation indicated that patient information was communicated to the receiving facility within 60 min of departure	Percentage of patients transferred to another healthcare facility whose medical record documentation indicated that physician information was communicated to the receiving facility within 60 min of departure	Percentage of patients transferred to another healthcare facility whose medical record documentation indicated that nursing information was communicated to the receiving facility within 60 min of departure	Patients who are transferred from an ED to another healthcare facility have communicated with the receiving facility within 60 min of discharge a list of tests done and results sent	Documents the extent to which a provider uses certified/qualified electronic health record (EHR) system that incorporates an electronic data interchange with one or more laboratories allowing for direct electronic transmission of laboratory data into the EHR as discrete searchable data elements
Measure title*	Vital Signs	Medication Information	Patient Information	Physician Information	Nursing Information	Procedures and Tests	The Ability for Providers with HIT to Receive Laboratory Data Electronically Directly into Their Qualified/Certified EHR System as Discrete Searchable Data Elements
NQF	0292	0293	0294	0295	0296	0297	0489

~		<i>a</i>				~	ntinued)
Ye	al Yes are, IQR	al Yes are,	al Yes are, IQR	Yes	Ye	Ye	(cor
	Hospit Compa	Hospit Comps OQR	Hospit Compa				
Endorsement removed (April 2014)	Endorsement removed (November 2018)	Endorsement removed (October 2018)	Endorsement removed (November 2018)	Endorsement removed (May 2012)	Endorsement removed (May 2012)	Endorsed	
2008	2008 ^{11,1}	2008 ^{41,4}	2008 ^{11,1}	2008 ^{11,1}	2008	2008	
oordination ure)	oordination ss)	oordination ss)	oordination ss)	oordination	oordination	osite	
Care c (struct	Care c (proce	Care c (proce	Care c (proce	Care c (other)	Care c (other)	Comp	
CMS	CMS	CMS	CMS	LSU	LSU	HEH	
Facility	Facility	Facility	Facility	Facility, clinician	Facility, clinician	Facility	
Documentation of the extent to which a provider uses a certified/qualified electronic health record (EHR) system to track pending laboratory tests, diagnostic studies (including common preventive screenings), or patient referrals	Median time from emergency department arrival to time of departure from the emergency room for patients admitted to the facility from the emergency department	Median time from emergency department arrival to time of departure from the emergency room for patients discharged from the emergency department (ED)	Median time from admit decision time to time of departure from the emergency department for emergency department patients admitted to inpatient status	Time of first contact in the ED to the time when the patient sees qualified medical personnel for patient evaluation and management	Percent of patients leaving without being seen by a qualified medical personnel	This measure focuses on adults 18 years and older with a diagnosis of severe sepsis or septic shock. Consistent with Surviving Sepsis Campaign guidelines, the measure contains several elements, including measurement of lactate, obtaining blood cultures, administering broad-spectrum antibiotics, fluid resuscitation, vasopressor administration, reassessment of volume status and tissue perfusion, and repeat lactate measurement	
Tracking of Clinical Results Between Visits	Median Time from ED Arrival to ED Departure for Admitted ED Patients	Median Time from ED Arrival to ED Departure for Discharged ED Patients	Admit Decision Time to ED Departure Time for Admitted Patients	Door to Diagnostic Evaluation by a Qualified Medical Personnel	Left Without Being Seen	Shock: Management Bundle	
0491	0495	0496	0497	0498	0499	0500	
(continued)							
-------------	--						
4.1							
Table							

Relevant to cancer care?§	oN	No	No	No	No	No	No
CMS program [‡]					Hospital Compare, OQR		
Current status	Endorsement removed (June 2012)	Endorsement removed (May 2012)	Endorsement removed (January 2013)	Endorsement removed (January 2013)	Endorsement removed (April 2017)	Endorsement removed (July 2012)	Endorsement removed (December 2013)
Year endorsed	2008 ¹¹⁴	2008 ^{II.9}	2008 ^{II.st}	2008 ^{11,1}		2009	2009
Measure target (type)*	Pulmonary (process)	Perinatal (process)	Cardiovascular (process)	Safety (process)	Musculoskeletal (process)	COPD (process)	Endocrine (process)
Measure owner ^{*,†}	CCF	ACEP	ACEP	AAP	CMS	NCQA	Optum
Reporting level	Population, integrated delivery system, facility, group, clinician	Facility, group, clinician	Facility, group, clinician	Facility	Population, facility	Population, health plan, integrated delivery system, facility, group, clinician	Population, health plan, integrated delivery system, facility, group, clinician
Measure description*	Any time an endotracheal tube is placed into a patient's airway in the emergency department (ED) or a patient arrives to the ED with an endotracheal tube already in place (via EMS or hospital transfer), there should be appropriate confirmation of ETT placement and documentation of its performance in the medical record	Percentage of female patients aged 14 to 50 who present to the emergency department (ED) with a chief complaint of abdominal pain for whom a pregnancy test ordered	Number of acute embolus patients who have orders for anticoagulation (heparin or low-molecular-weight heparin) for pulmonary embolus while in the ED	Percentage of emergency department visits by patients <18 years of age with a current weight documented in kilograms in the ED electronic health record; measure to be reported each month		This measure assesses the percentage of COPD exacerbations for members 40 years of age and older who had an acute inpatient discharge or ED encounter on or between January 1 and November 30 of the measurement year and who were dispensed appropriate medications	This measure identifies adults with diabetes mellitus that had a serum creatinine test in the last 12 reported months
Measure title*	Confirmation of Endotracheal Tube Placement	Pregnancy Test for Female Abdominal Pain Patients	Anticoagulation for Acute Pulmonary Embolus Patients	Pediatric Weight Documented in Kilograms	MRI Lumbar Spine for Low Back Pain	Pharmacotherapy Management of COPD Exacerbation (PCE)	Adult(s) with Diabetes Mellitus that Had a Serum Creatinine in the Last 12 Reported Months
NQF	0501	0502	0503	0504	0514	0549	0604

No	No	Yes	No	Ŷ	°N	oN
					Hospital Compare, OQR	Hospital Compare, OQR
Endorsement removed (December 2013)	Endorsement removed (March 2013)	Endorsement removed (July 2017)	Endorsement removed (July 2014)	Endorsement removed (November 2012)	Endorsed	Endorsement removed (January 2015)
2009	2010	2010	2011	2011	2011	2011
Cardiovascular (process)	Stroke (process)	Care coordination (process)	Perinatal (process)	(process)	Stroke (process)	Musculoskeletal (process)
Optum	Optum	AMA-PCPI	ACEP	CMS	CMS	CMS
Population, health plan, integrated delivery system, facility, group, clinician	Population, health plan, integrated delivery system, facility, group, clinician	Integrated delivery system, facility	Group, clinician	Facility	Facility	Facility
This measure identifies patients with hypertension (HTN) who had a serum creatinine in the last 12 reported months	Patient(s) with a recent emergency room encounter for a transient cerebral ischemic event that had any physician visit within 14 days of the acute event	Percentage of discharges from an emergency department (ED) to ambulatory care or home healthcare, in which the patient, regardless of age, or their caregiver(s), received a transition record at the time of ED discharge including, at a minimum, all of the specified elements	Percent of Rh-negative pregnant women at risk of fetal blood exposure who receive Rhogam in the ED	Emergency department acute myocardial infarction (AMI) patients or chest pain patients (with probable cardiac chest pain) with an order for troponin during the stay and having a time from ED arrival to completion of troponin results within 60 min of arrival	This measure calculates the percentage of acute ischemic stroke or hemorrhagic stroke patients who arrive at the ED within 2 h of the onset of symptoms and have a head computed tomography (CT) or magnetic resonance imaging (MRI) scan interpreted within 45 min of ED arrival	Median time from emergency department arrival to time of initial oral or parenteral pain medication administration for emergency department patients with a principal diagnosis of long bone fracture (LBF)
Patient(s) with Hypertension that Had a Serum Creatinine in the Last 12 Reported Months	Patients with a Transient Ischemic Event ER Visit that Had a Follow-Up Office Visit	Transition Record with Specified Elements Received by Discharged Patients (Emergency Department Discharges to Ambulatory Care [Home/Self-Care] or Home Health Care)	Rh immunoglobulin (Rhogam) for Rh-negative pregnant women at risk of fetal blood exposure	Troponin Results for Emergency Department Acute Myocardial Infarction (AMI) Patients or Chest Pain Patients (with Probable Cardiac Chest Pain) Received Within 60 min of Arrival	Head CT or MRI Scan Results for Acute Ischemic Stroke or Hemorrhagic Stroke Patients Who Received Head CT or MRI Scan Interpretation Within 45 Min of ED Arrival	Median Time to Pain Management for Long Bone Fracture
0605	0644	0649	0652	0660	0661	0662

(continued) 23

 Table 4.1 (continued)

Relevant to cancer	care? [§]	No	No	No	No	No	No	No
CMS	program [‡]							
	Current status	Endorsement removed (December 2013)	Endorsement removed (December 2013)	Endorsement removed (May 2017)	Endorsement removed (February 2016)	Endorsement removed (December 2015)	Endorsement removed (March 2015)	Endorsement removed (<i>February</i> 2014)
Year	endorsed	2011	2011	2011	2011	2011		2011
Measure target	(type)*	Cardiovascular (process)	Cardiovascular (process)	Pulmonary (process)	Pulmonary (efficiency)	Stroke (outcome)	Behavioral health (outcome)	Asthma (outcome)
Measure	owner*;†	Optum	Optum	ACEP	ACEP	HCI3	MGH	ALMA
	Reporting level	Population, health plan, integrated delivery system, facility, group, clinician	Population, health plan, integrated delivery system, facility, group, clinician	Group, clinician	Facility, group	Population, health plan, facility, group	Population, health plan, integrated delivery system, facility, group, clinician	Population, health plan
	Measure description*	This measure identifies patients with an emergency medicine visit for syncope that had an ECG done as part of their evaluation	This measure identifies patients with an emergency medicine visit for non-traumatic chest pain that had an ECG done as part of their evaluation	Percent of adult patients aged 18 years and older with an internal jugular central venous catheter placed in the emergency department (ED) under ultrasound guidance	Percent of patients undergoing CT pulmonary angiogram for the evaluation of possible PE who are at low-risk for PE consistent with guidelines(1, 2) prior to CT imaging	Percent of adult population aged 18–65 years who were admitted to a hospital with stroke, were followed for 1 month after discharge, and had one or more potentially avoidable complications (PACs)	The Pediatric Symptom Checklist (PSC) is a brief parent-report questionnaire that is used to assess overall psychosocial functioning in children from 3 to 18 years of age	Percentage of patients with asthma who have greater than or equal to one visit to the emergency room for asthma during the measurement period
	Measure title*	Patient(s) with an Emergency Medicine Visit for Syncope that Had an ECG	Patient(s) with an Emergency Medicine Visit for Non- traumatic Chest Pain that Had an ECG	Ultrasound Guidance for Internal Jugular Central Venous Catheter Placement	Inappropriate Pulmonary CT Imaging for Patients at Low Risk for Pulmonary Embolism	Proportion of Patients Hospitalized with Stroke that Have a Potentially Avoidable Complication (During the Index Stay or in the 30-Day Post-discharge Period)	Pediatric Symptom Checklist (PSC)	Asthma Emergency Department Visits
NOF	́Д	0664	0665	0666	0667	0705	0722	1381

Yes	Yes	No	Yes	Yes	Yes	(continued)
				Hospital Compare, IQR, VBP, HACRP, IRF QRP, LTCH QRP, PCHQR	Hospital Compare, IQR, VBP, HACRP, IRF QRP, LTCH QRP, PCHQR	Ŭ
Endorsed	Endorsed	Endorsement removed (July 2018)	Endorsement removed (December 2014)	Endorsed	Endorsed	
2012	2012	2012	2012	2012	2012	
Care coordination (cost/resource use)	Care coordination (cost/resource use)	Musculoskeletal (cost/resource use)	Pneumonia (cost/ resource use)	Infectious disease (outcome)	Infectious disease (outcome)	
HealthPartners	HealthPartners	Optum	Optum	CDC	CDC	
Population, group	Population, group	Population, health plan, integrated delivery system, facility, group, clinician	Population, health plan, integrated delivery system, facility, group, clinician	Population, facility	Population, facility	
The Resource Use Index (RUI) is a risk-adjusted measure of the frequency and intensity of services utilized to manage a provider group's patients. Resource use includes all resources associated with treating members including professional, facility inpatient and outpatient, pharmacy, lab, radiology, ancillary, and behavioral health services	Total Cost of Care reflects a mix of complicated factors such as patient illness burden, service utilization, and negotiated prices. Total Cost Index (TCI) is a measure of a primary care provider's risk-adjusted cost- effectiveness at managing the population they care for	The measure focuses on resources used to deliver episodes of care for patients who have undergone a hip/knee replacement	The measure focuses on resources used to deliver episodes of care for patients with pneumonia	Standardized infection ratio (SIR) and adjusted ranking metric (ARM) of hospital-onset unique blood source MRSA laboratory-identified events (LabID events) among all inpatients in the facility	Standardized infection ratio (SIR) and adjusted ranking metric (ARM) of hospital-onset CDI laboratory- identified events (LabID events) among all inpatients in the facility, excluding well-baby nurseries and neonatal intensive care units (NICUs)	
Total Resource Use Population- Based PMPM Index	Total Cost of Care Population- Based PMPM Index	ETG-Based Hip/Knee Replacement Cost of Care Measure	ETG-Based Pheumonia Cost of Care Measure	National Healthcare Safety Network (NHSN) Facility- Wide Inpatient Hospital-Onset Methicillin-Resistant <i>Staphylococcus aureus</i> (MRSA) Bacteremia Outcome Measure	National Healthcare Safety Network (NHSN) Facility- Wide Inpatient Hospital-Onset <i>Clostridium difficile</i> Infection (CDI) Outcome Measure	
1598	1604	1609	1611	1716	1717	

(continued)
4.1
Table

Relevant to cancer care?§	°Z	No	Yes	oZ	Yes	Yes
CMS program [‡]					ASCQR, Hospital Compare, OQR	
Current status	Endorsement removed (August 2016)	Endorsed	Endorsement removed (April 2017)	Endorsement removed (June 2020)	Endorsed	Endorsed
Year endorsed	2015	2012	2012	2014	2014	2015
Measure target (type)*	Asthma (process)	Asthma (process)	Disparities (process)	Care coordination (outcome)	Gastrointestinal (outcome)	Pediatric (outcome)
Measure owner*;†	NCQA	NCQA	GWU	CMS	CMS	AHRQ
Reporting level	Health plan, integrated delivery system	Health plan, integrated delivery system	Facility, group	Facility	Facility	Facility
Measure description [*]	The percentage of patients 5–64 years of age during the measurement year who were identified as having persistent asthma and were dispensed appropriate medications that they remained on during the treatment period. Two rates are reported. 1. The percentage of patients who remained on an asthma controller medication for at least 50% of their treatment period 2. The percentage of patients who remained on an asthma controller medication for at least 75% of their treatment period	The percentage of patients 5–64 years of age who were identified as having persistent asthma and had a ratio of controller medications to total asthma medications of 0.50 or greater during the measurement year	This measure is used to assess the percent of patient visits and admissions where preferred spoken language for healthcare is screened and recorded	Percentage of home health stays in which patients who had an acute inpatient hospitalization in the 5 days before the start of their home health stay used an emergency department but were not admitted to an acute care hospital during the 30 days following the start of the home health stay	Rate of risk-standardized, all-cause, unplanned hospital visits within 7 days of an outpatient colonoscopy among Medicare fee-for-service (FFS) patients aged 65 years and older	The Child Hospital Consumer Assessment of Healthcare Providers and Systems (Child HCAHPS) survey is a standardized survey instrument that asks parents and guardians (henceforth referred to as parents) of children under 18 years old to report on their and their child's experiences with innatient hosnital care
Measure title*	Medication Management for People with Asthma	Asthma Medication Ratio	L1A: Screening for Preferred Spoken Language for Healthcare	Emergency Department Use Without Hospital Readmission During the First 30 Days of Home Health	Facility 7-Day Risk- Standardized Hospital Visit Rate After Outpatient Colonoscopy	Child Hospital Consumer Assessment of Healthcare Providers and Systems (Child HCAHPS) Survey
NQF	1799	1800	1824	2505	2539	2548

°Z	Yes	No	°N	Yes	No	(continued)
Endorsed	Endorsed	Endorsed	Endorsed	Endorsed	Endorsed	
2015	2015	2015	2015	2015	2016	
Behavioral health (process)	Surgery (outcome)	Dental (outcome)	Dental (process)	Safety (outcome)	Behavioral health (process)	
ИСQА	CMS	ADA	ADA	HdYN	SCRI	
Population, health plan	Facility	Integrated delivery system	Integrated delivery system	Integrated delivery system, facility, group	Facility	
The percentage of discharges for patients 18 years of age and older who had a visit to the emergency department with a primary diagnosis of mental health or alcohol or other drug dependence during the measurement year <i>and</i> who had a follow-up visit with any provider with a corresponding primary diagnosis of mental health or alcohol or other drug dependence within 7 and 30 days of discharge	Facility-level, postsurgical risk- standardized hospital visit ratio (RSHVR) of the predicted to expected number of all-cause, unplanned hospital visits within 7 days of a same-day surgery at a hospital outpatient department (HOPD) among Medicare fee-for-service (FFS) patients aged 65 years and older	Number of emergency department visits for caries-related reasons per 100,000 member months for all enrolled children	Percentage of ambulatory care sensitive emergency department (ED) visits for dental caries among children 0–20 years in the reporting period for which the member visited a dentist within (a) 7 days and (b) 30 days of the ED visit	A Wrong-Patient Retract-and-Reorder (Wrong Patient-RAR) event occurs when an order is placed on a patient within an EHR, and is retracted within 10 min, and then the same clinician places the same order on a different patient within the next 10 min	Percentage of children/adolescents aged 5 to 19 years old seen in the emergency department with psychotic symptoms who are screened for alcohol or drugs of abuse	
Follow-Up After Emergency Department Visit for Mental Illness or Alcohol and Other Drug Abuse or Dependence	Hospital Visits After Hospital Outpatient Surgery	Ambulatory Care Sensitive Emergency Department Visits for Dental Caries in Children	Follow-Up After Emergency Department Visits for Dental Caries in Children	Wrong-Patient Retract-and- Reorder (Wrong Patient-RAR) Measure	Pediatric Psychosis: Screening for Drugs of Abuse in the Emergency Department	
2605	2687	2689	2695	2723	2806	

(continued)	
ble	
Ч	

Relevant to cancer	care? [§]	No	°Z	°Z	No	oN
CMS	$program^{\ddagger}$					
	Current status	Endorsed	Endorsed	Endorsed	Endorsed	Endorsed
Year	endorsed	2016	2016	2016	2016	2016
Measure target	(type)*	Stroke (process)	Stroke (process)	Stroke (process)	Cardiovascular (outcome)	Cardiovascular (outcome)
Measure	owner*;†	TJC	1J.C	1JC	CMS	CMS
	Reporting level	Facility	Facility	Facility	Facility	Facility
	Measure description*	Proportion of subarachnoid hemorrhage (SAH) patients aged 18 years and older for whom nimodipine treatment was administered within 24 h of arrival at this hospital	Proportion of ischemic stroke patients aged 18 years or older for whom an initial NIHSS score is performed prior to any acute recanalization therapy (i.e., intravenous (IV) thrombolytic (t-PA) therapy, or intra-arterial (IA) thrombolytic (t-PA) therapy, or mechanical endovascular reperfusion (MER) therapy) in patients undergoing recanalization therapy and documented in the medical record, or documented within 12 h of arrival at the hospital emergency department in patients who do not undergo recanalization therapy	Proportion of SAH and ICH stroke patients aged 18 years or older for whom a severity measurement (i.e., Hunt and Hess Scale for SAH patients or ICH Score for ICH patients) is performed prior to surgical intervention (e.g., clipping, coiling, or any surgical intervention) in patients undergoing surgical intervention and documented in the medical record, <i>or</i> documented within 6 h of arrival at the hospital emergency department in patients who do not undergo surgical intervention	The measure assesses days spent in acute care within 30 days of discharge from an inpatient hospitalization for HF to provide a patient-centered assessment of the post-discharge period	This measure assesses days spent in acute care within 30 days of discharge from an inpatient hospitalization for acute myocardial infarction (AMI) to provide a patient-centered assessment of the post-discharge period
	Measure title*	CSTK-06: Nimodipine Treatment Administered	CSTK-01: National Institutes of Health Stroke Scale (NIHSS) Score Performed for Ischemic Stroke Patients	CSTK-03: Severity Measurement Performed for Subarachnoid Hemorrhage (SAH) and Intracerebral Hemorrhage (ICH) Patients (Overall Rate)	Excess Days in Acute Care (EDAC) After Hospitalization for Heart Failure (HF)	Excess Days in Acute Care (EDAC) After Hospitalization for Acute Myocardial Infarction (AMI)
NQF	Ð	2863	2864	2866	2880	2881

8	s	0	0	s	S	0	intinued)
Ye	Ye	Ž	Ž	χ	Ye	Ž	(co
Endorsed	Approved for trial use (5)	Endorsed	Endorsed	Endorsed	Endorsed	Endorsed	
2016	2017	2019	2018	2018	2019	2018	
Pulmonary (outcome)	Safety (outcome)	Cardiovascular (Outcome)	Safety (process)	Surgery (outcome)	Urology (outcome)	Safety (process)	
CMS	CCF	АНА	CMS	CMS	CMS	CMS	
Facility	Facility	Facility	Facility	Facility	Facility	Population	
This measure assesses days spent in acute care within 30 days of discharge from an inpatient hospitalization for pneumonia, including aspiration pneumonia or for sepsis (not severe sepsis) with a secondary discharge diagnosis of pneumonia coded in the claim as present on admission (POA) and no secondary diagnosis of severe sepsis coded as POA	Percentage of laboratory potassium samples drawn in the emergency department (ED) with hemolysis	This measure estimates a hospital -level risk standardized survival rate (RSSR) for patients aged 18 years and older who experience an inhospital cardiac arrest	Patients aged 18 years and older prescribed two or more opioids or an opioid and benzodiazepine concurrently at discharge from a hospital-based encounter (inpatient or emergency department [ED], including observation stays)	Facility-level risk-standardized ratio of acute, unplanned hospital visits within 7 days of a general surgery procedure performed at an ambulatory surgical center (ASC) among Medicare fee-for-service (FFS) patients aged 65 years and older	Facility-level risk-standardized rate of acute, unplanned hospital visits within 7 days of a urology procedure performed at an ambulatory surgical center (ASC) among Medicare fee-for-service (FFS) patients aged 65 years and older	The percentage of Medicaid beneficiaries ages 18–64 with an OUD who filled a prescription for or were administered or dispensed an FDA-approved medication for the disorder during the measure year	
Excess Days in Acute Care (EDAC) After Hospitalization for Pneumonia	Potassium Sample Hemolysis in the Emergency Department	Risk-Standardized Survival Rate (RSSR) for Inhospital Cardiac Arrest	. Safe Use of Opioids— Concurrent Prescribing	Facility-Level 7-Day Hospital Visits After General Surgery Procedures Performed at Ambulatory Surgical Centers	Hospital Visits After Urology Ambulatory Surgical Center Procedures	Use of Pharmacotherapy for Opioid Use Disorder (OUD)	
2882	29836	3309	3316	3357	3366	3400	

 Table 4.1 (continued)

Relevant to cancer care? [§]	No	°Z	Yes	No	Ŷ
CMS program [‡]					
Current status	Endorsed	Endorsed	Endorsed	Endorsed	Endorsed
Year endorsed	2019	2019	2019	2019	2019
Measure target (type)*	Safety (process)	(process)	Musculoskeletal (outcome)	Behavioral health (process)	Behavioral health (process)
Measure owner ^{*,†}	CMS	IMPAQ	CMS	NCQA	NCQA
Reporting level	Population	Health plan	Facility	Health plan	Health plan
Measure description [*]	Percentage of discharges from an inpatient or residential treatment for substance use disorder (SUD) for Medicaid beneficiaries, ages 18 to 64, which was followed by a treatment service for SUD	The percentage of issuer-product-level acute events requiring either an emergency department (ED) visit or hospitalization for one of the following six chronic conditions: hypertension, asthma, heart failure (HF), coronary artery disease (CAD), chronic obstructive pulmonary disease (COPD), or diabetes mellitus (Type I or Type II), where follow-up was received within the timeframe recommended by clinical practice guidelines in a nonemergency outpatient setting	Facility-level risk-standardized rate of acute, unplanned hospital visits within 7 days of an orthopedic procedure performed at an ambulatory surgical center (ASC) among Medicare fee-for-service (FFS) patients aged 65 years and older	The percentage of emergency department (ED) visits for members 13 years of age and older with a principal diagnosis of alcohol or other drug (AOD) abuse or dependence, who had a follow-up visit for AOD	The percentage of emergency department (ED) visits for members 6 years of age and older with a principal diagnosis of mental illness or intentional self-harm, who had a follow-up visit for mental illness
Measure title*	Continuity of Care After Inpatient or Residential Treatment for Substance Use Disorder (SUD)	Timely Follow-Up After Acute Exacerbations of Chronic Conditions	Hospital Visits After Orthopedic Ambulatory Surgical Center Procedures	Follow-Up After Emergency Department Visit for Alcohol and Other Drug Abuse or Dependence	Follow-Up After Emergency Department Visit for Mental Illness
NQF	3453	3455	3470	3488	3489

Relevance to cancer care is based on the authors' analysis of the measure specifications. Some measures (e.g., NQF #0090-Emergency Medicine: 12-Lead Electrocardiogram (ECG) Performed Measure was approved for trial use by NQF. NQF's Trial Use Program enables temporary use of electronic measures, or eMeasures, that lack full reliability and validity testing but are otherwise for Non-traumatic Chest Pain) may apply to cancer patients with other comorbidities as well as health status changes. However, the measures are not directly relevant to cancer care Medicare/Quality-Initiatives-Patient-Assessment-Instruments/Value-Based-Programs/HVBP/Hospital-Value-Based-Purchasing) and the set of the set

[¶]Measure was given time-limited endorsement. Time-limited endorsement is granted to measures that meet NQF evaluation criteria, but have not been adequately field-tested. Measure developers are given up to 2 years to demonstrate the reliability, validity, and feasibility of the measure based on testing at multiple provider sites [219] ready for implementation [218]

including the IQR program and PPS-Exempt Cancer Hospitals Quality Reporting Program (PCHQR). Over time, many of these measures have been retired from federal reporting programs or are no longer endorsed by the NQF. As of July 2020, there are 39 NQF-endorsed ED quality measures, including 9 ED quality measures used in CMS reporting programs. One additional ED quality measure has been approved for trial use (see Table 4.1). ED measures relevant to cancer care and the limitations of those measures are summarized in the following section and in Table 4.2.

Table 4.2 Existing ED measures relevant to cancer care, current gaps, and measure development priorities

Cancer-specific ED measures

Description: Measure aspects of emergency care that are unique to cancer patients. Include measures of the processes, outcomes, structure, efficiency, and costs of care as well as patients' perception-of-care

Rationale: Cancer patients visit the ED throughout the continuum of care and often present with complex, interrelated symptom burden (particularly at the end of life). Many ED measures focus on cardiovascular disease and are not relevant to oncologic emergency care. In addition, many cancer patients experience unique quality of care issues (e.g., late-stage cancers presenting to the ED) that reflect quality issues in other care settings and are not captured in existing measures. Widespread adoption of cancer-specific ED measures will help stimulate improvements in emergency oncologic care

Current measures: Two cancer-specific ED measures have been developed, and one measure is currently National Quality Forum (NQF)endorsed. They assess overutilization of ED services, due to poor symptom management, aggressive treatment, poor care coordination, or inadequate access to care

Examples:

- NQF measure #3490—Admission and Emergency Department (ED) Visits for Patients Receiving Outpatient Chemotherapy Health services research priorities: Listed below by measure type

Measure development priorities: Listed below by measure type

ED outcome measures

Description: Measure the outcomes of emergency care, including the sustainability of health post-ED discharge, timeliness of ED care, and treatment complications during and after ED discharge

Rationale: Cancer patients frequently visit the ED for symptom management (e.g., management of acute pain and fatigue) due to cancer treatment or cancer progression. In addition, ED care delays are associated with ED overcrowding and boarding and, ultimately, poorer outcomes and compromised quality of life. Failure to measure the timeliness of care—in particular, timely symptom improvement—represents a failure to measure the most important outcomes for these patients

Current measures: There are 17 NQF-endorsed ED outcome measures; these measures largely assess overutilization of ED services and patient safety

Examples:

- NQF measure #2882-Excess days in acute care (EDAC) after hospitalization for pneumonia
- NQF measure #3309-Risk-Standardized Survival Rate (RSSR) for Inhospital Cardiac Arrest

Health services research priorities:

- Develop protocols to adopt validated patient-reported outcome surveys as a standard of care for EDs to collect data on symptom burden and quality of life in the ED and post-ED discharge. Focus on minimizing patient burden and leveraging telehealth and other emerging technologies, where possible
- Study clinical and patient characteristics that are associated with repeat ED visits and post-ED discharge health decline in the cancer population

Measure development priorities:

- "Time to" patient-reported symptom improvement in the ED, stratified by chief complaint
- "Time to" cancer diagnosis, for patients presenting to the ED with an undiagnosed cancer
- Sustainability of patient-reported symptom improvement post-ED discharge, stratified by chief complaint
- Repeat ED visits within 2, 7, and 14 days of ED discharge, stratified by chief complaint
- ED length of stay for cancer patients, stratified by (1) patients admitted to an inpatient unit, (2) patients transferred to another facility, and (3) patients discharged home

ED process measures

Description: Assess compliance with established standards of ED care that have been linked to improved patient outcomes, fewer unnecessary services, and more equitable care. Include a wide array of measures, such as adherence to guideline-based diagnostic testing and treatment; protocols around patient intake, discharge, and care coordination; and policies to ensure equitable care for vulnerable patient populations *Rationale*: Routine measurement of adherence to guideline-based care can highlight practice variations that contribute to poorer outcomes and higher costs of care. Measuring care coordination by ED providers is important to ensure that patients are guided to appropriate follow-up care and to prevent repeat ED visits and inpatient admissions

Current measures: There are 19 NQF-endorsed ED process measures. Fourteen of these measures are condition- or population-specific; several measures focus on substance use disorder. Two ED process measures evaluate care coordination for patients discharged to outpatient care *Examples*:

- NQF measure #0004-Initiation and Engagement of Alcohol and Other Drug Abuse or Dependence Treatment

- NQF measure #0291—Emergency Transfer Communication Measure

Table 4.2 (continued)

Health services research priorities:

- Develop algorithms to identify patients at potential risk of presenting to the ED with an undiagnosed cancer
- Test methods to promote care coordination between outpatient oncologic and ED providers
- Model episodes of oncologic emergency care, with well-defined endpoints and treatment pathways
- Develop algorithms to identify patients at risk for disparities in care that present to the ED

Measure development priorities:

- Screening and diagnosis for high-frequency complications that present to the ED (e.g., pain, fatigue, dyspnea, nausea, dehydration,
- depression, and cognitive impairment)
- Patients discharged with a referral to an appropriate outpatient oncologic provider
- Advance care planning discussions for patients with advanced cancer
- Cancers diagnosed in the ED, stratified by (1) cancer type and (2) stage of disease

ED cost /resource use measures

Description: Calculate direct and indirect costs (or assess resource utilization) for a specific healthcare service, episode of care, or medical condition. Demonstrate variations in costs across geographic regions, medical conditions, care delivery settings, and providers

Rationale: Cost and resource use measures can increase transparency around cost inefficiencies (perceived and actual), higher costs associated with adverse events, delayed diagnosis and treatment, and individual patient factors, such as comorbid conditions. Furthermore, these measures can provide important insights into cost variation between providers, care delivery settings, and geographic regions, among patients with similar diagnoses, and across the continuum of cancer care

Current measures: There are two NQF-endorsed cost/resource use measures that are relevant to ED care

Examples

- NQF measure #1598-Total Resource Use Population-Based PMPM Index

- NQF measure #1604-Total Cost of Care Population-Based PMPM Index

Health services research priorities:

- Model episodes of oncologic emergency care, with well-defined endpoints and treatment pathways

Measure development priorities:

- Costs of care per ED visit, stratified by chief complaint
- Cost of diagnosing asymptomatic or quasi-symptomatic cancers in the ED
- Costs of managing patient comorbidities in the ED
- Costs of care by adverse event
- Costs of ED care in the last 7, 14, and 30 days of life

ED efficiency measures

Description: Examine the relationship between inputs and outputs in emergency care; they compare resource use (and associated costs) with the level of health outcome achieved

Rationale: Significant resources are expended in managing the complex—and often interrelated—symptoms, comorbidities, and psychosocial needs of patients presenting to the ED, particularly cancer patients

Current measures: There are no NQF-endorsed ED efficiency measures

Examples: None

Health services research priorities:

- Understand the overuse, underuse, and misuse of ED resources in cancer patients; this is largely unstudied beyond the frequency of ED visits. Develop guidelines for appropriate ED resource utilization for cancer patients
- Evaluate the relationship between ED resource utilization and outcomes for cancer patients
- Study the relationship between resource utilization (in the ED and in the outpatient setting) and repeat ED visits for cancer patients. Develop protocols to reduce repeat ED visits for cancer patients, particularly at the end of life

Measure development priorities:

- Efficient utilization of advanced imaging studies for cancer patients

ED patients' perception-of-care measures

Description: Evaluate patients' satisfaction or experience with the healthcare received

Rationale: While restoration of health is of primary importance among cancer patients, equally important is patient (and caregiver) experience with care throughout the cancer care continuum. This is particularly true for patients with advanced cancer whose treatment may be noncurative.

Current measures: One ED patients' perception-of-care survey has been developed, but it is not NQF-endorsed or used in public reporting programs. The Child Hospital Consumer Assessment of Healthcare Providers and Systems (Child HCAHPS) Survey is NQF-endorsed but classified as an outcome measure

Examples:

- Consumer Assessment of Healthcare Providers and Systems (CAHPS) survey for emergency care (ED CAHPS) *Health services research priorities*:

- Strategies to address the psychosocial needs of cancer patients with advanced disease and their caregivers

Measure development and research priorities:

- Relevance of the ED CAHPS survey to oncologic emergency medicine

- Survey of caregiver experience with emergency care and overall caregiver burden

Source: This table is based on the authors' analysis of NQF-endorsed ED measures relevant to cancer care [220], current gaps, and measure development priorities as of July 2020

Limitations of Existing Quality Measures for Emergency Departments

Despite the noteworthy efforts to date, the current ED performance measurement landscape has substantial limitations. For example, several ED measures are reported on Hospital Compare. However, these measures are largely provideroriented, reflect fragmented care delivery, and lack a clear method to address upstream care delivery challenges that often present in the ED. Moreover, there is no nationally mandated public reporting program for emergency care. Hence, patients lack reliable guidance on ED provider performance. Due to these factors, current reporting efforts offer limited potential to improve substantially the quality of ED care for cancer patients. Five limitations of ED quality measurement in the USA are briefly discussed below: (1) gaps in existing ED measures; (2) fragmented measure development; (3) difficulty defining an episode of oncologic emergency care; (4) measurement without a clear mechanism for improving ED care; and (5) challenges in obtaining ED quality data.

Gaps in Existing ED Measures

A robust ED measure set for cancer patients should evaluate multiple dimensions of oncologic emergency care, including access, care coordination, advance care planning, patient and family engagement, and evaluation and management of acute and chronic conditions, as well as psychosocial needs. Routine measurement of outcomes, costs of care, and appropriate resource utilization is also essential. Yet, no existing measure set or quality reporting program adequately measures these aspects of oncologic emergency care. As noted above, 31 ED quality measures are endorsed or approved for trial use by the NOF as of July 2020. Eighteen of these measures are relevant to cancer care, including one cancerspecific measure. Current ED measurement gaps relevant to cancer care span measure categories-process, outcomes, cost/resource use, efficiency, and patients' perception-ofcare-and include cancer-specific ED measures. These measurement gaps, together with recommendations to address these gaps, are summarized in Table 4.2.

Fragmented Measure Development

Historically, ED measure development efforts in the USA were academic-led and focused on specific patient populations or clinical conditions. These initiatives were conducted independently of payers and state and federal agencies, leading to a "patchwork of measures" for ED care [208]. These independent measure development efforts have contributed to the fragmented ED quality measurement observed today, which undermines efforts to deliver high-quality, patient-centered

care. With few exceptions, existing ED measures have not been widely adopted by providers or payers. Thus, most ED care is not routinely measured, and existing quality measures provide an incomplete view of the nation's ED system. A wellcoordinated approach to developing ED quality of care measures for oncologic is discussed later in this chapter.

Difficulty Defining an Episode of Oncologic Emergency Care

Defining an episode of emergency care is challenging for most conditions due to variations in the expected prognosis, treatment time, time to recovery, care teams, and treatment settings. Defining standardized episodes of oncologic emergency care is especially problematic for two reasons. First, cancer patients move frequently and unpredictably between care settings throughout the continuum of care. Therefore, cancer patients may present to the ED before diagnosis (for late-stage cancers presenting to the ED), at any point during treatment, and at the end of life. Second, the sequelae of cancer and its treatment vary greatly across patients. Therefore, cancer patients can present to the ED with symptoms of varying severity, ranging from moderate dehydration to lifethreatening sepsis, making it difficult to standardize oncologic emergency treatment pathways across patients. Because episodes of oncologic emergency care can vary so greatly across patients, it is equally difficult to develop quality measures and appropriate benchmarks for care. Focused health service research is needed to develop episodes of oncologic emergency care with well-defined endpoints to support the development of relevant quality measures for this setting.

Measurement Without a Clear Mechanism for Improving Care

Quality measures designed for performance improvement and accountability should align with evidence-based guidelines, should be actionable by clinicians, and have a clearly defined relationship with patient outcomes. Moreover, measures should be reported publicly to inform healthcare consumers and to drive improvements in care. Public reporting of ED performance data has been proposed as a critical lever for improving the nation's emergency care system [35]. Ideally, existing quality measurement programs could be leveraged to measure the quality of emergency care across providers and care settings. However, the current programs are too narrowly focused to support a broad, system-level approach to measuring the quality of emergency care. Additionally, experience with publicly reported ED measures has produced mixed results. Some public reporting initiatives (e.g., AMI performance measures) have generated significant improvements in care,

while others (e.g., pneumonia performance measures) have yielded disappointing results or—even worse—poorer quality of care. In those cases, the measures were misaligned with the existing guidelines, were based on weak evidence, or included arbitrary time points [208]. Additionally, flawed attribution models may generate erroneous conclusions regarding the quality of ED care and impede efforts to address quality of care issues. These limitations must be addressed to ensure that ED quality measures can support meaningful improvements in care, particularly for oncologic where multiple providers and care settings share responsibility for their outcomes of care.

Challenges in Obtaining ED Quality of Care Data

Much has been published in recent years regarding the limitations of existing data to support robust, actionable quality measurement. Historical quality measurement leveraged administrative claims data, which are relatively easy to access but are not designed for quality reporting. At best, they offer an incomplete view of healthcare quality. At worst, the data may be irrelevant, incomplete, or inaccurate. Federal agencies and EHR vendors have promoted EHRs as a viable alternative to address these data issues. However, EHRs were designed to support healthcare operations, rather than quality measurement, and early assessments of EHR-based quality reporting have been disappointing [221-223]. Manual chart review and data entry remains a primary method of collecting dataor supplementing electronic data-to produce quality measures. The level of effort and reporting lag associated with this approach limit access to the data that are critical for timely, actionable, and meaningful ED quality measurement. Moreover, because ED physicians often lack an established and ongoing relationship with their patients, they often lack access to immediate and long-term data on the outcomes of ED care. Potential strategies to address these issues are described later in this chapter.

Desired State of National Quality Measurement for Oncologic Emergency Care

In reviewing the history and current state of national quality measurement for emergency medicine, several important themes emerge:

- 1. There is widespread acknowledgment of the essential role that EDs serve in the nation's public health system.
- 2. Quality issues in emergency medicine are well documented, and healthcare experts have developed practical recommendations to address many of these issues.

- Some quality issues observed in the ED are unrelated to the quality of emergency care and, instead, reflect broader social issues (e.g., inadequate access to healthcare) or quality of care issues in other healthcare settings.
- 4. Public and private organizations have recognized that quality measurement is integral to ED quality improvement, and early successes in cardiovascular emergency medicine have demonstrated how ED-based national quality measurement can be leveraged to improve patient outcomes.
- HIT advancements, together with increased adoption of EHRs, offer the potential to give ED providers greater access to the data needed to care for their patients and to evaluate their quality of care on a more real-time basis.

While not specific to oncologic emergency care, these accomplishments represent a solid platform to address existing measurement gaps through national reporting for oncologic emergency care. There are five factors that contribute to the current state of inertia we described earlier in this chapter: (1) gaps in existing ED measures; (2) fragmented measure development; (3) difficulty defining the episode of oncologic emergency care; (4) measurement without a clear mechanism for improving ED care; and (5) challenges in obtaining ED quality data. Many of these factors stem from substantial shortcomings in the funding, oversight, and coordination of measure development and public reporting for cancer care.

In this section, we outline a vision for measuring quality in oncologic emergency care, through the implementation of the NAM's recommendation to create a comprehensive national quality reporting program for cancer care. This includes a well-coordinated approach to developing cancerspecific ED quality of care measures. We also propose healthcare policy changes that will promote better alignment between public reporting and reimbursement for oncologic emergency care and that will promote shared accountability across providers. Additionally, we describe how the NAM's recommendation to implement a learning healthcare system for cancer could address many of the challenges in obtaining ED quality of care data. Finally, we share a sample case study describing how ED and palliative providers can partner with palliative care providers to address upstream gaps in care to reduce ED visits.

Vision for National Quality Measurement in Oncologic Emergency Care

Since 1999, the NAM has promoted national quality measurement as an essential lever to improve the quality of US cancer care delivery. In 2013, the NAM released *Delivering High-Quality Cancer Care: Charting a New Course for a System in Crisis*, which outlined six components of a high-quality cancer care delivery system: (1) engaged patients; (2) an adequately staffed, trained, and coordinated workforce; (3) evidence-based cancer care; (4) a learning healthcare information technology system; (5) translation of evidence into clinical practice, quality measurement, and performance improvement; and (6) accessible, affordable cancer care [84]. The report identified the nation's inability to systematically measure and improve cancer care delivery as a primary contributor to existing gaps in cancer quality and offered the following recommendation:

Recommendation 8: Quality Measurement

Goal: Develop a national quality reporting program for cancer care as part of a learning healthcare system.

To accomplish this, the Department of Health and Human Services should work with professional societies to:

- Create and implement a formal long-term strategy for publicly reporting quality measures for cancer care that leverages existing efforts
- Prioritize, fund, and direct the development of meaningful quality measures for cancer care with a focus on outcome measures and with performance targets for use in publicly reporting the performance of institutions, practices, and individual clinicians
- Implement a coordinated, transparent reporting infrastructure that meets the needs of all stakeholders, including patients, and is integrated into a learning healthcare system [84]

Implementation of this national quality reporting program for cancer care would enhance quality measurement within the ED and across other care delivery settings. It would support purposeful, well-coordinated, and patient-centered quality measurement in the ED, with an emphasis on care coordination and shared accountability across providers and care delivery settings. Through public reporting, it would encourage evidence-based care delivery and patient engagement while discouraging unnecessary-and potentially harmful-care. By increasing transparency around the outcomes, processes, and costs of cancer care, the national reporting program envisioned in the report could expedite progress toward a high-quality cancer care delivery system, of which the ED is an essential component. Adequate funding, formal leadership, strong collaboration, and HIT enhancements, together with a well-developed framework and a unified strategy, are essential to its successful implementation, as discussed below.

Health Policy for Measuring Quality in Oncologic Emergency Care

As described earlier in this chapter, EMTALA and the no-duty-to-treat principle form the health policy base for emergency care in the USA. While EMTALA facilitates patient access to emergency medical care, it does not regulate the quality of that care. More recently, the MMA, DRA, Tax Relief and Health Care Act of 2006, and MACRA introduced and reformed national quality reporting for emergency care. The quality reporting stimulated by this legislation has done little to promote high-quality oncologic emergency care because it focused largely on other conditions, such as cardiovascular disease.

To advance quality in the nation's oncologic emergency care, national quality reporting for cancer care is essential, as recommended by the NAM. The frequency, complexity, and costs of oncologic emergency care, particularly at the end of life, necessitate a well-coordinated and unified approach to address current measurement gaps in oncologic emergency care. Thus, we offer the following policy recommendations in support of this effort:

- Leadership and Collaboration: Delivering High-Quality Cancer Care—Charting a New Course for a System in Crisis identified HHS as the appropriate organizer of this work. Through collaboration with patient advocacy organizations, professional societies, payers, and other stakeholders, HHS could ignite national development of quality measures for oncologic emergency care. Designating CMS and the NQF as key partners in this effort could accelerate progress in developing validated cancer-specific ED quality of care measures.
- *Formal Long-Term Strategy*: Create and enforce a formal long-term strategy (with shorter-term milestones) and a well-defined framework for the development and public reporting of measures for oncologic emergency care (as part of a broader strategy and framework for cancer). This long-term strategy would address the needs of all cancer patients, with a particular focus on cancer patients seeking emergency care at the end of life. Moreover, by moving away from quality measurement focused on specific Medicare payment programs, it would promote shared accountability by providers.
- *Research*: Fund health services research and clinical trials to expand the scientific evidence for oncologic emergency care, including:
 - Effective care coordination between outpatient oncologic and ED providers
 - Outpatient care delivery models that reduce unnecessary ED utilization among cancer patients

- Approaches to mitigate the overutilization of ED services by cancer patients, particularly at the end of life
- Episodes of oncologic emergency care, with welldefined endpoints and treatment pathways
- Strategies to address the psychosocial needs of cancer patients with advanced disease and their caregivers
- Drivers of late-stage cancers presenting to the ED
- Care delivery models that integrate palliative care with ED services
- Measure Development: Fund the development of a robust set of meaningful measures for oncologic emergency care (including performance targets) for use in public reporting. Measure development should focus on the outcomes of care as well as access to care, care coordination, advance care planning, patient and family engagement, and evaluation and management of acute and chronic conditions, as well as psychosocial needs. High-priority measurement gaps are described in Table 4.2. Prioritization of measure development should align with the formal longterm strategy guiding this effort and target likely healthcare disparities. Moreover, measure development should have a well-defined cost-benefit relationship and should foster shared accountability across providers and including patients. Where appropriate, the developed measures should address multiple care delivery settings. Measures available from existing data sources should receive higher priority. However, lack of data should not constitute a barrier to measure development. A formal tool should be developed to assist the collaborative in prioritizing measure development [224].
- *Transparent Reporting Infrastructure*: As recommended by the NAM, implement a reporting infrastructure (including IT infrastructure and reporting methodologies) that promotes transparency of the outcomes that are most meaningful to patients and their caregivers and that meets the information needs of all stakeholders (patients and their caregivers, providers, payers, and state and federal agencies). Public reporting should be understandable by patients and their caregivers to support healthcare decision-making.

Expedited adoption of health policy in support of these priorities would do much to address the existing measurement gaps for oncologic emergency care. With multistakeholder collaboration among organizations that share a vested interest in oncologic emergency medicine as well as proper funding and authority, robust national quality measurement for oncologic emergency care could become a reality within a few years.

HIT Support Through the Learning Healthcare System for Cancer

Providers face significant obstacles in obtaining timely, actionable, and comprehensive data to support the robust quality measurement described herein. Additionally, because ED providers lack an established and ongoing relationship with their patients, they often do not have access to postdischarge and longitudinal outcomes data to support meaningful quality measurement. To advance meaningful quality measurement and public reporting, Delivering High-Quality Cancer Care: Charting a New Course for a System in Crisis advocated the creation of a learning healthcare system for cancer [84]. A learning healthcare system streamlines provider data collection and reporting and enables real-time data analysis for performance improvement, quality measurement, and clinical decision support. The cancer-specific learning healthcare system described by the NAM would support more rapid innovation in cancer care delivery by addressing critical data gaps in two ways: (1) by capturing provider-driven clinical data, patient-reported outcomes, and patient and caregiver experience with care in a structured format and (2) by integrating structured, unstructured, and semi-structured data. National endorsement of this recommendation would address many of the data gaps described in this report and would enable development and reporting of quality measures for oncologic emergency care. To be successful, federal incentives to promote HIT adoption (e.g., Meaningful Use) should incorporate the principles of a learning healthcare system for cancer [225]. Likewise, public and private payers should reward providers for participating in a learning healthcare system for cancer. Aligning provider incentives with adoption of a learning healthcare system for cancer would enhance the current IT infrastructure and promote widespread access to the information needed to catalyze national public reporting for oncologic emergency care.

Role of Targeted Quality Measures in Driving Practice Change

As noted earlier in this chapter, quality measures provide a standardized, objective means of evaluating healthcare quality and hold an important role in the US healthcare delivery system. State and federal agencies utilize quality measures to promote provider accountability and to inform the public. Increasingly, payers are using quality measures in valuebased payment programs to align reimbursement with quality of care. Because cancer patients experience unique quality of care issues and because most disease-specific ED measures focus on non-cancer conditions, the existing ED quality of care measures offer minimal opportunity to improve the quality of oncologic emergency care. Despite these limitations, *appropriately selected* quality measures have the potential to inform consumer decision-making and care planning, accelerate improvements in care, and highlight variation between providers and over time within a given practice setting [84]. Additionally, routine quality measurement and reporting enables payers and providers to test whether new care delivery and payment models have a positive effect on the accessibility, quality, and affordability of healthcare.

Public reporting of well-designed quality measures for oncologic emergency care represents a powerful policy lever to encourage more appropriate ED resource utilization, better care coordination, shared accountability, and, ultimately, superior outcomes and patient (and caregiver) experience with care. Lamb et al. observed that the act of measuring performance at the provider level can ignite an interest in selfimprovement or a competitive spirit among providers, leading to improvements in care [226]. Pay-for-performance programs are another policy lever that could lead to improvements in the quality of oncologic emergency care. The effectiveness of pay-for-performance programs has been debated extensively, given current measurement gaps across multiple conditions and various aspects of care. However, designing a pay-for-performance program around targeted quality measures for oncologic emergency care (such as those listed as measure development priorities in Table 4.2) could stimulate significant and lasting improvements in care.

Conclusion

In this chapter, we examined the history, current state, and desired future state of health policy for quality in oncologic emergency care. We discussed five quality issues that cancer patients experience when seeking care in the ED, together with upstream drivers. We also described specific issues for dedicated oncologic EDs. We highlighted the essential role of quality measures in addressing these quality of care issues, along with five limitations of the existing quality measures that apply to emergency care. We also shared the quality measures for emergency care that are currently endorsed by the NQF and used in CMS quality reporting programs. We outlined recommendations for national quality measurement for oncologic emergency care, through the implementation of the NAM's recommendation to create national quality reporting for cancer care, as part of a learning healthcare system. We proposed health policy changes-in the form of leadership and collaboration, formal long-term strategy, research, measure development, and transparent reporting

infrastructure—to accelerate progress toward national quality measurement for oncologic emergency care. We emphasized the importance of adequate funding, formal leadership, strong collaboration, and HIT enhancements to make this reporting a reality. We also explained how a learning healthcare system for cancer and targeted quality measures can catalyze change and advance progress toward the national reporting program described herein. Additionally, we shared a sample case study outlining a collaborative approach to address unmet palliative care needs in the ED.

The recommendations outlined in this chapter are ambitious but are necessary to accelerate the development of targeted quality measures for oncologic emergency medicine. To be successful, measure developers and other stakeholders must abandon the historical practice of siloed development of highly specific measures that apply to a small proportion of the population or to a single care delivery setting. With adequate funding, unified leadership, and multi-stakeholder commitment, national quality reporting for oncologic emergency medicine could become a reality within a few years, leading to more patient-centered and higher-quality cancer care in the ED.

References

- Reineck LA, Le TQ, Seymour CW, Barnato AE, Angus DC, Kahn JM. Effect of public reporting on intensive care unit discharge destination and outcomes. Ann Am Thorac Soc. 2015;12(1):57–63.
- Ketelaar NA, Faber MJ, Flottorp S, Rygh LH, Deane KH, Eccles MP. Public release of performance data in changing the behaviour of healthcare consumers, professionals or organisations. Cochrane Database Syst Rev. 2011;(11):CD004538.
- Metcalfe D, Rios Diaz AJ, Olufajo OA, et al. Impact of public release of performance data on the behaviour of healthcare consumers and providers. Cochrane Database Syst Rev. 2018;9:CD004538.
- 4. Rosenbaum S. The enduring role of the emergency medical treatment and active labor act. Health Affair. 2013;32(12):2075–81.
- 5. Hurley v. Eddingfield, 59 N.E. 1058 (Ind. 1901).
- Birmingham Baptist Hospital v. Crews, 157 So.152d 224 (Ala. 1934).
- 7. Campbell v. Mincey, 413 F. Supp. 416, 418 (N.D. Miss. 1975).
- 8. Childs v. Weis, 440 S.W.442d 104 (Ct Civ App Tx. 1969).
- 9. Ricks v Budge, 91 Utah 307, 364 P302d 208 (Utah 1937).
- 10. Mead v Adler, Or App 451, 220 P453d 118 (Or 2009).
- Hospital Survey and Construction Act. In. Vol Pub. L. No. 79-725, 60 Stat. 1040, codified as amended, 42 US Code, sec. 291. 1946.
- Manlove v Wilmington General Hospital, 53 Del. 339, 169 A332d
 Super 1961), affd 1954 Del 1915, 1174 A1962d 1135 (Del 1961).
- Doe v. Bridgeton Hospital Association, Inc., 71 N. J. 478 (1976), cert. den. 1433 U.S. 1914, 1997 S.Ct. 2987, 1953 L.Ed. 1972d 1100 (N. J. 1977).
- Mercy Medical Center of Oshkosh, Inc. v Winnebago County, 58 Wis.52d 260, 206 N.W.262d 198 (Wis 1973).
- Rev. Rul. 69-545, 1969-2 C.B. 117, modified by, Rev. Rul. 83-157, 1983-2 C.B. 94.
- 16. 42 CFR, sec. 124.603(b).

- Americans with Disabilities Act of 1990. I:. Vol Pub. L. No. 101-336, 104 Stat. 327, codified as 42 US Code, sec. 12182. 1990.
- Civil Rights Act of 1964, Title VI. In. Vol Pub. L. No. 88-352, 78 Stat. 252, codified as amended as 42 US Code, sec. 2000d(a). 1964.
- Consolidated Omnibus Budget Reconciliation Act of 1985. In. Vol Pub. L. No. 99-272, 100 Stat. 82, codified as amended, 42 US Code, sec. 1395d(d). 1986.
- Centers for Medicare & Medicaid Services. 1135 waivers. December 2017. https://www.cms.gov/Medicare/Provider-Enrollment-and-Certification/SurveyCertEmergPrep/1135-Waivers. Accessed 18 Jul 2020.
- Centers for Medicare & Medicaid Services. Waiver or modification of requirements under section 1135 of the social security act. March 13, 2020. https://www.phe.gov/emergency/news/healthactions/section1135/Pages/covid19-13March20.aspx. Accessed 18 Jul 2020.
- 22. 42 CFR, sec. 489.24(b), 74 FR 44001. 2009.
- 23. Harry v. Marchant, 291 F. 293d 767 (211th Cir. 2002), en banc.
- Power v. Arlington Hospital Association, 42 F. 43d 851 (4th Cir. 1994).
- 25. Summers v Baptist Medical Center Arkadelphia, 69 F63d 902 (908th Cir 1995), rev on reh 1991 F1993d 1132 (1996).
- 26. Matter of Baby K, 16 F.13d 590 (594th Cir. 1994), cert. denied, 1115 S. Ct. 1991, 1130 L. Ed. 1992d 1942 (1994).
- 27. 68 Fed. Reg. 53229. 2003.
- Patient Protection and Affordable Care Act. In. Vol Pub. L. No. 111-148, 124 Stat. 119 §9007. 2010.
- Patient Protection and Affordable Care Act. In. Vol Pub. L. No. 111-148, 124 Stat. 119 §1303. 2010.
- Rosenbaum S. The impact of United States law on medicine as a profession. JAMA. 2003;289(12):1546–56.
- 31. 42 CFR, sec. 489.24(d)(2), 68 Fed. Reg. 53222. 2003.
- McKenna RM, Purtle J, Nelson KL, Roby DH, Regenstein M, Ortega AN. Examining EMTALA in the era of the patient protection and Affordable Care Act. AIMS Public Health. 2018;5(4):366–77.
- American College of Emergency Physicians. The impact of unreimbursed care on the emergency physician. 2020. https://www.acep. org/administration/reimbursement/the-impact-of-unreimbursedcare-on-the-emergency-physician. Accessed 17 Jul 2020.
- 34. Bitterman RA. EMTALA: The law that forever changed the practice of EM. ACEP Now. 2018. https://www.acepnow.com/ article/emtala-the-law-that-forever-changed-the-practice-of-em/. Accessed 17 Jul 2020.
- Institute of Medicine. Hospital-based emergency care: at the breaking point. Washington: The National Academies Press; 2007. https://doi.org/10.17226/11621.
- 36. Sikka V, Ornato JP. Cancer diagnosis and outcomes in Michigan EDs vs other settings. Am J Emerg Med. 2012;30(2):283–92.
- Amri R, Bordeianou LG, Sylla P, Berger DL. Colon cancer surgery following emergency presentation: effects on admission and stageadjusted outcomes. Am J Surg. 2015;209(2):246–53.
- Livingood WC, Smotherman C, Lukens-Bull K, Aldridge P, Kraemer DF, Wood DL, Volpe C. An elephant in the emergency department: symptom of disparities in cancer care. Popul Health Manag. 2016;19(2):95–101.
- Solsky I, Rapkin B, Wong K, Friedmann P, Muscarella P, In H. Gastric cancer diagnosis after presentation to the ED: the independent association of presenting location and outcomes. Am J Surg. 2018;216(2):286–92.
- 40. Weithorn D, Arientyl V, Solsky I, Umadat G, Levine R, Rapkin B, Leider J. Diagnosis setting and colorectal cancer outcomes: the impact of cancer diagnosis in the emergency department. J Surg Res. 2020;255:164–71.
- Thompson CA, Sheridan P, Murphy JD, Lyratzopoulos G. Abstract PR10: emergency department-mediated cancer diagnosis in the United States. J Clin Oncol. 2020;29(6 Suppl 2):PR10.

- Duflos C, Antoun S, Loirat P, DiPalma M, Minvielle E. Identification of appropriate and potentially avoidable emergency department referrals in a tertiary cancer care center. Support Care Cancer. 2017;25(8):2377–85.
- 43. Wallace EM, Cooney MC, Walsh J, Conroy M, Twomey F. Why do palliative care patients present to the emergency department? Avoidable or unavoidable? Am J Hosp Palliat Care. 2013;30(3):253–6.
- 44. Delgado-Guay MO, Kim YJ, Shin SH, Chisholm G, Williams J, Allo J, Bruera E. Avoidable and unavoidable visits to the emergency department among patients with advanced cancer receiving outpatient palliative care. J Pain Symptom Manag. 2015;49(3):497–504.
- 45. Barbera L, Atzema C, Sutradhar R, Seow H, Howell D, Husain A, et al. Do patient-reported symptoms predict emergency department visits in cancer patients? A population-based analysis. Ann Emerg Med. 2013 Apr;61(4):427–437.e5.
- 46. Henson LA, Gao W, Higginson IJ, Smith M, Davies JM, Ellis-Smith C, Daveson BA. Emergency department attendance by patients with cancer in their last month of life: a systematic review and meta-analysis. J Clin Oncol. 2015;33(4):370–6.
- 47. Langton JM, Blanch B, Drew AK, Haas M, Ingham JM, Pearson SA. Retrospective studies of end-of-life resource utilization and costs in cancer care using health administrative data: a systematic review. Palliat Med. 2014;28(10):1167–96.
- Carlson MD, Herrin J, Du Q, Epstein AJ, Barry CL, Morrison RS, Back AL, Bradley EH. Impact of hospice disenrollment on health care use and medicare expenditures for patients with cancer. J Clin Oncol. 2010;28(28):4371–5.
- 49. Earle CC, Park ER, Lai B, Weeks JC, Ayanian JZ, Block S. Identifying potential indicators of the quality of end-of-life cancer care from administrative data. J Clin Oncol. 2003;21(6):1133–8.
- Earle CC, Neville BA, Landrum MB, Souza JM, Weeks JC, Block SD, et al. Evaluating claims-based indicators of the intensity of end-of-life cancer care. Int J Qual Health Care. 2005;17(6):505–9.
- 51. Aprile G, Pisa FE, Follador A, Foltran L, De Pauli F, Mazzer M, et al. Unplanned presentations of cancer outpatients: a retrospective cohort study. Support Care Cancer. 2013;21(2):397–404.
- 52. Lurie N, Margolis GS, Rising KL. The US emergency care system: meeting everyday acute care needs while being ready for disasters. Health Aff (Millwood). 2013;32(12):2166–71.
- Salehi L, Phalpher P, Valani R, et al. Emergency department boarding: a descriptive analysis and measurement of impact on outcomes. CJEM. 2018;20(6):929–37.
- Singer AJ, Thode HC Jr, Viccellio P, Pines JM. The association between length of emergency department boarding and mortality. Acad Emerg Med. 2011;18(12):1324–9.
- 55. Felton BM, Reisdorff EJ, Krone CN, Laskaris GA. Emergency department overcrowding and inpatient boarding: a statewide glimpse in time. Acad Emerg Med. 2011;18(12):1386–91.
- Fogarty E, Saunders J, Cummins F. The effect of boarders on emergency department process flow. J Emerg Med. 2014;46(5):706–10.
- Kulstad EB, Sikka R, Sweis RT, Kelley KM, Rzechula KH. ED overcrowding is associated with an increased frequency of medication errors. Am J Emerg Med. 2010;28(3):304–9.
- Shen YC, Hsia RY. Association between ambulance diversion and survival among patients with acute myocardial infarction. JAMA. 2011;305(23):2440–7.
- Yankovic N, Glied S, Green LV, Grams M. The impact of ambulance diversion on heart attack deaths. Inquiry. 2010;47(1):81–91.
- Mariotto AB, Yabroff KR, Shao Y, Feuer EJ, Brown ML. Projections of the cost of cancer care in the United States: 2010–2020. J Natl Cancer Inst. 2011;103(2):117–28. Erratum in: J Natl Cancer Inst. 2011;103(8):699.
- Vera-Llonch M, Weycker D, Glass A, Gao S, Borker R, Barber B, Oster G. Healthcare costs in patients with metastatic lung cancer receiving chemotherapy. BMC Health Serv Res. 2011;11:305.

- 62. Vera-Llonch M, Weycker D, Glass A, Gao S, Borker R, Qin A, Oster G. Healthcare costs in women with metastatic breast cancer receiving chemotherapy as their principal treatment modality. BMC Cancer. 2011;11:250.
- 63. Hu YY, Kwok AC, Jiang W, Taback N, Loggers ET, Ting GV, Lipsitz SR, Weeks JC, Greenberg CC. High-cost imaging in elderly patients with stage IV cancer. J Natl Cancer Inst. 2012;104(15):1164–72.
- 64. Goodman DC, Fisher ES, Chang CH, Morden NE, Jacobson JO, Murray K, Miesfeldt S. Quality of end-of-life cancer care for Medicare beneficiaries: regional and hospital specific analyses. Lebanon: The Dartmouth Institute for Health Policy and Clinical Practice; 2010. http://www.dartmouthatlas.org/downloads/reports/ Cancer_report_11_16_10.pdf. Accessed 6 Sept 2020.
- 65. Bekelman JE, Halpern SD, Blankart CR, Bynum JP, Cohen J, Fowler R, International Consortium for End-of-Life Research (ICELR), et al. Comparison of site of death, health care utilization, and hospital expenditures for patients dying with cancer in 7 developed countries. JAMA. 2016;315(3):272–83.
- 66. Haque W, Verma V, Butler EB, Teh BS. Patterns of end-of-life oncologic care for stage iv non-small cell lung cancer in the United States. Anticancer Res. 2019;39(6):3137–40.
- 67. (CHCF) California HealthCare Foundation. Final chapter: Californians' attitudes and experiences with death and dying. 2012. https://www.chcf.org/publication/final-chapter-californians-attitudesand-experiences-with-death-and-dying/. Accessed 6 Sept 2020.
- National Journal. The Regence Foundation. Living well at the end of life: a national conversation. 2011. http://syndication.nationaljournal.com/communications/NationalJournalRegenceToplin.pdf. Accessed 6 Sept 2020.
- 69. Wennberg JE, Fisher ES, Goodman DC, Skinner JS. Tracking the course of patients with severe chronic illness. Lebanon: The Dartmouth Institute for Health Policy and Clinical Practice; 2020. http://www.dartmouthatlas.org/downloads/atlases/2008_Chronic_ Care_Atlas.pdf. Accessed 6 Sept 2020.
- Institute of Medicine. Variation in health care spending: target decision making, not geography. Washington: The National Academies Press; 2013. https://doi.org/10.17226/18393.
- 71. Smith AK, Fisher J, Schonberg MA, Pallin DJ, Block SD, Forrow L, et al. Am I doing the right thing? Provider perspectives on improving palliative care in the emergency department. Ann Emerg Med. 2009;54(1):86–93, 93.e1.
- Varney SM, Vargas TE, Pitotti RL, Bebarta VS. Reasons military patients with primary care access leave an emergency department waiting room before seeing a provider. South Med J. 2012;105(10):538–42.
- Taylor C, Benger JR. Patient satisfaction in emergency medicine. Emerg Med J. 2004;21(5):528–32.
- 74. Boudreaux ED, Friedman J, Chansky ME, Baumann BM. Emergency department patient satisfaction: examining the role of acuity. Acad Emerg Med. 2004;11(2):162–8.
- 75. Welch SJ. Twenty years of patient satisfaction research applied to the emergency department: a qualitative review. Am J Med Qual. 2010;25(1):64–72.
- Sonis JD, Aaronson EL, Lee RY, Philpotts LL, White BA. Emergency department patient experience: a systematic review of the literature. J Patient Exp. 2018;5(2):101–6.
- 77. Abidova A, da Silva PA, Moreira S. Predictors of patient satisfaction and the perceived quality of healthcare in an emergency department in Portugal. West J Emerg Med. 2020;21(2):391–403.
- Toma G, Triner W, McNutt LA. Patient satisfaction as a function of emergency department previsit expectations. Ann Emerg Med. 2009;54(3):360–7, e366.
- McCarthy DM, Ellison EP, Venkatesh AK, Engel KG, Cameron KA, Makoul G, Adams JG. Emergency department team communication with the patient: the patient's perspective. J Emerg Med. 2013;45(2):262–70.

- Boudreaux ED, Ary RD, Mandry CV, McCabe B. Determinants of patient satisfaction in a large, municipal ED: the role of demographic variables, visit characteristics, and patient perceptions. Am J Emerg Med. 2000;18(4):394–400.
- Hansagi H, Carlsson B, Brismar B. The urgency of care need and patient satisfaction at a hospital emergency department. Health Care Manag Rev. 1992;17(2):71–5.
- McMillan JR, Younger MS, DeWine LC. Satisfaction with hospital emergency department as a function of patient triage. Health Care Manag Rev. 1986;11(3):21–7.
- Emergency Department Consumer Assessment of Healthcare Providers and Systems (ED CAHPS®) Survey, Version 1.0. Recommended guidelines. Baltimore: Centers for Medicare & Medicaid Services; 2020. Accessed 6 Sept 2020.
- Institute of Medicine. Delivering high-quality cancer care: charting a new course for a system in crisis. Washington: The National Academies Press; 2013. https://doi.org/10.17226/18359.
- Reinhard SC, Levine C. Home alone: family caregivers providing complex chronic care. Washington: AARP Public Policy Institute; 2012. https://www.aarp.org/content/dam/aarp/research/public_ policy_institute/health/home-alone-family-caregivers-providingcomplex-chronic-care-rev-AARP-ppi-health.pdf. Accessed July 29, 2020.
- Stenberg U, Ruland CM, Miaskowski C. Review of the literature on the effects of caring for a patient with cancer. Psychooncology. 2010;19(10):1013–25.
- Braun M, Mikulincer M, Rydall A, Walsh A, Rodin G. Hidden morbidity in cancer: spouse caregivers. J Clin Oncol. 2007;25(30):4829–34.
- Grunfeld E, Coyle D, Whelan T, Clinch J, Reyno L, Earle CC, et al. Family caregiver burden: results of a longitudinal study of breast cancer patients and their principal caregivers. CMAJ. 2004;170(12):1795–801.
- Wright AA, Keating NL, Balboni TA, Matulonis UA, Block SD, Prigerson HG. Place of death: correlations with quality of life of patients with cancer and predictors of bereaved caregivers' mental health. J Clin Oncol. 2010;28(29):4457–64.
- Wadhwa D, Burman D, Swami N, Rodin G, Lo C, Zimmermann C. Quality of life and mental health in caregivers of outpatients with advanced cancer. Psychooncology. 2013;22(2):403–10.
- Mazanec SR, Daly BJ, Douglas SL, Lipson AR. Work productivity and health of informal caregivers of persons with advanced cancer. Res Nurs Health. 2011;34(6):483–95.
- Cameron JI, Franche RL, Cheung AM, Stewart DE. Lifestyle interference and emotional distress in family caregivers of advanced cancer patients. Cancer. 2002;94(2):521–7.
- Institute of Medicine. Cancer care for the whole patient: meeting psychosocial health needs. Washington: The National Academies Press; 2008. https://doi.org/10.17226/11993.
- 94. Segrin C, Badger TA, Meek P, Lopez AM, Bonham E, Sieger A. Dyadic interdependence on affect and quality-of-life trajectories among women with breast cancer and their partners. J Soc Pers Relat. 2005;22(5):673–89.
- 95. Segrin C, Badger T, Dorros SM, Meek P, Lopez AM. Interdependent anxiety and psychological distress in women with breast cancer and their partners. Psycho-Oncology. 2007;16(7):634–43.
- 96. Recognize, Assist, Include, Support, and Engage (RAISE) Family Caregiver Act of 2017. In. Pub. L. No. 115-119, 132 Stat. 23. 2018.
- AARP. Politics & society/advocacy. support caregivers. State law to help family caregivers. 2014. https://www.aarp.org/politics-society/advocacy/caregiving-advocacy/info-2014/aarp-createsmodel-state-bill.html. Accessed 21 Jul 2020.
- Jadalla A, Ginex P, Coleman M, Vrabel M, Bevans M. Family Caregiver strain and burden: a systematic review of evidence-based interventions when caring for patients with cancer. Clin J Oncol Nurs. 2020;24(1):31–50.

- 99. Sun V, Grant M, Koczywas M, Freeman B, Zachariah F, Fujinami R, et al. Effectiveness of an interdisciplinary palliative care intervention for family caregivers in lung cancer. Cancer. 2015;121(20):3737–45.
- 100. Hendrix CC, Bailey DE Jr, Steinhauser KE, Olsen MK, Stechuchak KM, Lowman SG, et al. Effects of enhanced caregiver training program on cancer caregiver's self-efficacy, preparedness, and psychological well-being. Support Care Cancer. 2016;24(1):327–36.
- 101. Grant M, Sun V, Fujinami R, Sidhu R, Otis-Green S, Juarez G, et al. Family caregiver burden, skills preparedness, and quality of life in non-small cell lung cancer. Oncol Nurs Forum. 2013;40(4):337–46.
- Wikipedia. NCI-designated cancer center. https://en.wikipedia. org/wiki/NCI-designated_Cancer_Center. Accessed 10 Sept 2020.
- National Cancer Institute. NCI-Designated Cancer Centers: Find a Cancer Center. http://www.cancer.gov/researchandfunding/extramural/cancercenters/find-a-cancer-center. Accessed 9 Sept 2020.
- 104. Diaz v. Division of Social Services, 628 S.E.622d 621 (N.C. 2006).
- 105. Sussman J, Baldwin LM. The interface of primary and oncology specialty care: from diagnosis through primary treatment. J Natl Cancer Inst Monogr. 2010;2010(40):18–24.
- Nekhlyudov L, Latosinsky S. The interface of primary and oncology specialty care: from symptoms to diagnosis. J Natl Cancer Inst Monogr. 2010;2010(40):11–7.
- 107. Grunfeld E, Earle CC. The interface between primary and oncology specialty care: treatment through survivorship. J Natl Cancer Inst Monogr. 2010;2010(40):25–30.
- Taplin SH, Clauser S, Rodgers AB, Breslau E, Rayson D. Interfaces across the cancer continuum offer opportunities to improve the process of care. J Natl Cancer Inst Monogr. 2010;2010(40):104–10.
- 109. Taplin SH, Rodgers AB. Toward improving the quality of cancer care: addressing the interfaces of primary and oncology-related subspecialty care. J Natl Cancer Inst Monogr. 2010;2010(40):3–10.
- Hussain T, Chang HY, Veenstra CM, Pollack CE. Fragmentation in specialist care and stage III colon cancer. Cancer. 2015;121(18):3316–24.
- 111. Chobanuk J. Medication errors and shift to a culture of patient safety and high reliability. Can Oncol Nurs J. 2016;26(3):255.
- Siddiqui S, Cruz I. A cancer patient journey: complete review during acute treatment phase. Health Equity. 2019;3(1):403–8.
- Weaver SJ, Jacobsen PB. Cancer care coordination: opportunities for healthcare delivery research. Transl Behav Med. 2018;8(3):503–8.
- 114. van Ryn M, Phelan SM, Arora NK, Haggstrom DA, Jackson GL, Zafar SY, et al. Patient-reported quality of supportive care among patients with colorectal cancer in the Veterans Affairs Health Care System. J Clin Oncol. 2014;32(8):809–15.
- 115. Gandhi TK, Bartel SB, Shulman LN, Verrier D, Burdick E, Cleary A, et al. Medication safety in the ambulatory chemotherapy setting. Cancer. 2005;104(11):2477–83.
- 116. Walsh KE, Dodd KS, Seetharaman K, Roblin DW, Herrinton LJ, Von Worley A, et al. Medication errors among adults and children with cancer in the outpatient setting. J Clin Oncol. 2009;27(6):891–6.
- 117. Shooshtari A, Kalidindi Y, Jung J. Cancer care spending and use by site of provider-administered chemotherapy in Medicare. Am J Manag Care. 2019;25(6):296–300.
- 118. Tsui J, Howard J, O'Malley D, Miller WL, Hudson SV, Rubinstein EB, et al. Understanding primary care-oncology relationships within a changing healthcare environment. BMC Fam Pract. 2019;20(1):164.
- Gorin SS, Haggstrom D, Han PKJ, Fairfield KM, Krebs P, Clauser SB. Cancer care coordination: a systematic review and meta-

analysis of over 30 years of empirical studies. Ann Behav Med. 2017;51(4):532–46.

- 120. Geerse OP, Lamas DJ, Bernacki RE, Sanders JJ, Paladino J, Berendsen AJ, et al. Adherence and concordance between serious illness care planning conversations and oncology clinician documentation among patients with advanced cancer. J Palliat Med. 2020; https://doi.org/10.1089/jpm.2019.0615.
- 121. Green MJ, Van Scoy LJ, Foy AJ, Dimmock AEF, Lehman E, Levi BH. Patients with advanced cancer choose less aggressive medical treatment on vignettes after using a computer-based decision aid. Am J Hosp Palliat Care. 2020;37(7):537–41.
- 122. National Comprehensive Cancer Network. NCCN clinical practice guidelines in oncology: palliative care. Version 1. 2018. https://oncolife.com.ua/doc/nccn/Palliative_Care.pdf. Accessed 9 Sept 2020.
- 123. Mack JW, Cronin A, Keating NL, Taback N, Huskamp HA, Malin JL, et al. Associations between end-of-life discussion characteristics and care received near death: a prospective cohort study. J Clin Oncol. 2012;30(35):4387–95.
- 124. McDermott CL, Engelberg RA, Sibley J, Sorror ML, Curtis JR. The association between chronic conditions, end-of-life health care use, and documentation of advance care planning among patients with cancer. J Palliat Med. 2020; https://doi.org/10.1089/ jpm.2019.0530.
- 125. Chen CH, Chou WC, Chen JS, Chang WC, Hsieh CH, Wen FH, Tang ST. An individualized, interactive, and advance care planning intervention promotes transitions in prognostic awareness states among terminally ill cancer patients in their last six monthsa secondary analysis of a randomized controlled trial. J Pain Symptom Manage. 2020;60(1):60–9, e6.
- 126. Mack JW, Weeks JC, Wright AA, Block SD, Prigerson HG. Endof-life discussions, goal attainment, and distress at the end of life: predictors and outcomes of receipt of care consistent with preferences. J Clin Oncol. 2010;28(7):1203–8.
- 127. Detering KM, Hancock AD, Reade MC, Silvester W. The impact of advance care planning on end of life care in elderly patients: randomised controlled trial. BMJ. 2010;340:c1345.
- 128. Green MJ, Schubart JR, Whitehead MM, Farace E, Lehman E, Levi BH. Advance care planning does not adversely affect hope or anxiety among patients with advanced cancer. J Pain Symptom Manag. 2015;49(6):1088–96.
- 129. Cohen MG, Althouse AD, Arnold RM, Bulls HW, White D, Chu E, et al. Is advance care planning associated with decreased hope in advanced cancer? JCO Oncol Pract. 2020; https://doi. org/10.1200/OP.20.00039.
- Keating NL, Landrum MB, Rogers SO Jr, Baum SK, Virnig BA, Huskamp HA, et al. Physician factors associated with discussions about end-of-life care. Cancer. 2010;116(4):998–1006.
- 131. Mack JW, Cronin A, Taback N, Huskamp HA, Keating NL, Malin JL, et al. End-of-life care discussions among patients with advanced cancer: a cohort study. Ann Intern Med. 2012;156(3):204–10.
- 132. Ishihara KK, Wrenn K, Wright SW, Socha CM, Cross M. Advance directives in the emergency department: too few, too late. Acad Emerg Med. 1996;3(1):50–3.
- 133. Tan TS, Jatoi A. End-of-life hospital costs in cancer patients: do advance directives or routes of hospital admission make a difference? Oncology. 2011;80(1–2):118–22.
- 134. Narang AK, Wright AA, Nicholas LH. Trends in advance care planning in patients with cancer: results from a national longitudinal survey. JAMA Oncol. 2015;1(5):601–8.
- 135. Goswami P, Mistric M, Diane BF. Advance care planning: advanced practice provider-initiated discussions and their effects on patient-centered end-of-life care. Clin J Oncol Nurs. 2020;24(1):81–7.
- 136. Pajka SE, Hasdianda MA, George N, Sudore R, Schonberg MA, Bernstein E, et al. Feasibility of a brief intervention to facilitate

advance care planning conversations for patients with life-limiting illness in the emergency department. J Palliat Med. 2020; https://doi.org/10.1089/jpm.2020.0067.

- 137. Greer JA, Jackson VA, Meier DE, Temel JS. Early integration of palliative care services with standard oncology care for patients with advanced cancer. CA Cancer J Clin. 2013;63(5):349–63.
- 138. Bandieri E, Banchelli F, Artioli F, Mucciarini C, Razzini G, Cruciani M, et al. Early versus delayed palliative/supportive care in advanced cancer: an observational study. BMJ Support Palliat Care. 2019;10(4):e32. https://doi.org/10.1136/ bmjspcare-2019-001794.
- 139. Temel JS, Greer JA, Admane S, Gallagher ER, Jackson VA, Lynch TJ, et al. Longitudinal perceptions of prognosis and goals of therapy in patients with metastatic non-small-cell lung cancer: results of a randomized study of early palliative care. J Clin Oncol. 2011;29(17):2319–26.
- 140. Morrison RS, Dietrich J, Ladwig S, Quill T, Sacco J, Tangeman J, Meier DE. Palliative care consultation teams cut hospital costs for Medicaid beneficiaries. Health Aff (Millwood). 2011;30(3):454–63.
- 141. Chang HT, Lin MH, Chen CK, Chen TJ, Tsai SL, Cheng SY, et al. Medical care utilization and costs on end-of-life cancer patients: the role of hospice care. Medicine (Baltimore). 2016;95(44):e5216.
- 142. Temel JS, Greer JA, Muzikansky A, Gallagher ER, Admane S, Jackson VA, et al. Early palliative care for patients with metastatic non-small-cell lung cancer. N Engl J Med. 2010;363(8):733–42.
- 143. Coates A, Porzsolt F, Osoba D. Quality of life in oncology practice: prognostic value of EORTC QLQ-C30 scores in patients with advanced malignancy. Eur J Cancer. 1997;33(7):1025–30.
- 144. Djarv T, Metcalfe C, Avery KN, Lagergren P, Blazeby JM. Prognostic value of changes in health-related quality of life scores during curative treatment for esophagogastric cancer. J Clin Oncol. 2010;28(10):1666–70.
- 145. Mehanna HM, Morton RP. Does quality of life predict long-term survival in patients with head and neck cancer? Arch Otolaryngol Head Neck Surg. 2006;132(1):27–31.
- 146. Fang FM, Liu YT, Tang Y, Wang CJ, Ko SF. Quality of life as a survival predictor for patients with advanced head and neck carcinoma treated with radiotherapy. Cancer. 2004;100(2):425–32.
- 147. Hui D, Elsayem A, De la Cruz M, Berger A, Zhukovsky DS, Palla S, et al. Availability and integration of palliative care at US cancer centers. JAMA. 2010;303(11):1054–61.
- 148. Morrison RS, Augustin R, Souvanna P, Meier DE. America's care of serious illness: a state-by-state report card on access to palliative care in our nation's hospitals. J Palliat Med. 2011;14(10):1094–6.
- 149. Hui D, De La Rosa A, Chen J, Dibaj S, Delgado Guay M, et al. State of palliative care services at US cancer centers: an updated national survey. Cancer. 2020;126(9):2013–23.
- 150. America's care of serious illness: a state-by-state report card on access to palliative care in our nation's hospitals. New York: Center to Advance Palliative Care and the National Palliative Care Research Center; 2019. https://reportcard.capc.org/wp-content/ uploads/2020/05/CAPC_State-by-State-Report-Card_051120.pdf
- 151. Yu JA, Ray KN, Park SY, Barry A, Smith CB, Ellis PG, Schenker Y. System-level factors associated with use of outpatient specialty palliative care among patients with advanced cancer. J Oncol Pract. 2019;15(1):e10–e19.4.
- 152. Barth C, Colombet I, Montheil V, Huillard O, Boudou-Rouquette P, Tlemsani C, et al. First referral to an integrated onco-palliative care program: a retrospective analysis of its timing. BMC Palliat Care. 2020;19(1):31.
- 153. Limehouse WE, Feeser VR, Bookman KJ, Derse A. A model for emergency department end-of-life communications after acute devastating events-part I: decision-making capacity, surrogates, and advance directives. Acad Emerg Med. 2012;19(9):E1068–72.

- 154. Limehouse WE, Feeser VR, Bookman KJ, Derse A. A model for emergency department end-of-life communications after acute devastating events-part II: moving from resuscitative to end-of-life or palliative treatment. Acad Emerg Med. 2012;19(11):1300–8.
- 155. Chan G, Bryant EN, Lamba S, Weissman DE, Quest TE, Knox TH. Clinical practice guidelines: a technical assistance resource from the IPAL-EM project. New York: The IPAL-EM Project, Center to Advance Palliative Care; 2011. Available from: http:// ipal.capc.org/downloads/ipal-em-clinical-practice-guidelines.pdf.
- 156. Wilson JG, English DP, Owyang CG, Chimelski EA, Grudzen CR, Wong HN, Aslakson RA, AAHPM Research Committee Writing Group. End-of-life care, palliative care consultation, and palliative care referral in the emergency department: a systematic review. J Pain Symptom Manage. 2020;59(2):372–83, e1.
- 157. Reuter Q, Marshall A, Zaidi H, Sista P, Powell ES, McCarthy DM, Dresden SM. Emergency department-based palliative interventions: a novel approach to palliative care in the emergency department. J Palliat Med. 2019;22(6):649–55.
- 158. da Silva SD, Nunes CM, Gomes B. Effectiveness of emergency department based palliative care for adults with advanced disease: a systematic review. J Palliat Med. 2016;19(6):601–9.
- 159. Rosenberg M, Rosenberg L. Integrated model of palliative care in the emergency department. West J Emerg Med. 2013;14(6):633–6.
- 160. Grudzen CR, Richardson LD, Major-Monfried H, Kandarian B, Ortiz JM, Morrison RS. Hospital administrators' views on barriers and opportunities to delivering palliative care in the emergency department. Ann Emerg Med. 2013;61(6):654–60.
- 161. Grudzen CR, Richardson LD, Hopper SS, Ortiz JM, Whang C, Morrison RS. Does palliative care have a future in the emergency department? Discussions with attending emergency physicians. J Pain Symptom Manag. 2012;43(1):1–9.
- 162. George N, Phillips E, Zaurova M, Song C, Lamba S, Grudzen C. Palliative care screening and assessment in the emergency department: a systematic review. J Pain Symptom Manag. 2016;51(1):108–19. e102
- 163. Wright R, Hayward B, Kistler E, Vaughan Dickson V, Grudzen C. That was a game changer: clinical impact of an emergency department-based palliative care communication skills training workshop. Emerg Med J. 2020;37(5):262–4.
- 164. Quest TE, Asplin BR, Cairns CB, Hwang U, Pines JM. Research priorities for palliative and end-of-life care in the emergency setting. Acad Emerg Med. 2011;18(6):e70–6.
- 165. Shoenberger J, Lamba S, Goett R, DeSandre P, Aberger K, Bigelow S, et al. Development of hospice and palliative medicine knowledge and skills for emergency medicine residents: using the Accreditation Council for Graduate Medical Education Milestone Framework. AEM Educ Train. 2018;2(2):130–45.
- 166. George N, Bowman J, Aaronson E, Ouchi K. Past, present, and future of palliative care in emergency medicine in the USA. Acute Med Surg. 2020;7(1):e497.
- 167. Kraus CK, Greenberg MR, Ray DE, Dy SM. Palliative care education in emergency medicine residency training: a survey of program directors, associate program directors, and assistant program directors. J Pain Symptom Manag. 2016;51(5):898–906.
- Aberger K, Wang D. Palliative care for frontline physicians: case studies highlighting best practices. Springer Nature: Cham; 2020.
- 169. Boyle S. Integrating palliative care in the emergency department: a paradigm shift. New York: Center to Advance Palliative Care; 2019. https://www.capc.org/blog/palliative-pulse-the-palliative-pulse-december-2018-integrating-palliative-care-in-the-emergency-department-a-paradigm-shift/. Accessed 23 Jul 2020.
- 170. Integrating palliative care practices in the emergency department. New York: Center to Advance Palliative Care; 2020. https://www. capc.org/toolkits/integrating-palliative-care-practices-in-theemergency-department/. Accessed July 23, 2020.

- 171. Lamba S, DeSandre PL, Todd KH, Bryant EN, Chan GK, Grudzen CR, Improving Palliative Care in Emergency Medicine Board, et al. Integration of palliative care into emergency medicine: the Improving Palliative Care in Emergency Medicine (IPAL-EM) collaboration. J Emerg Med. 2014;46(2):264–70, 4.
- 172. Kelley AS, Deb P, Du Q, Aldridge Carlson MD, Morrison RS. Hospice enrollment saves money for Medicare and improves care quality across a number of different lengths-of-stay. Health Aff (Millwood). 2013;32(3):552–61.
- 173. Bentur N, Resnizky S, Balicer R, Eilat-Tsanani T. Utilization and cost of services in the last 6 months of life of patients with cancer – with and without home hospice care. Am J Hosp Palliat Care. 2014;31(7):723–5.
- 174. Tax Equity and Fiscal Responsibility Act of 1982. In. Vol Pub. L. No. 97-248, 96 Stat. 324 §122. 1982.
- 175. Medicare Payment Advisory Commission. Hospice services. Chapter 12. In: Report to the congress: medicare payment policy. 2020. http://medpac.gov/docs/default-source/reports/mar20_medpac_ch12_sec.pdf?sfvrsn=0. Accessed 9 Sept 2020.
- 176. 80 Fed. Reg. 47142. 2015.
- 177. Wright AA, Katz IT. Letting go of the rope–aggressive treatment, hospice care, and open access. N Engl J Med. 2007;357(4):324–7.
- 178. Byock I, Twohig JS, Merriman M, Collins K. Promoting excellence in end-of-life care: a report on innovative models of palliative care. J Palliat Med. 2006;9(1):137–51.
- Krakauer R, Spettell CM, Reisman L, Wade MJ. Opportunities to improve the quality of care for advanced illness. Health Aff (Millwood). 2009;28(5):1357–9.
- 180. Mor V, Joyce NR, Coté DL, Gidwani RA, Ersek M, Levy CR, et al. The rise of concurrent care for veterans with advanced cancer at the end of life. Cancer. 2016;122(5):782–90.
- 181. Presley CJ, Han L, O'Leary JR, Zhu W, Corneau E, Chao H, et al. Concurrent hospice care and cancer-directed treatment for advanced lung cancer and receipt of aggressive care at the end of life in the Veteran's Health Administration. J Palliat Med. 2020;23(8):1038–44.
- 182. Mor V, Wagner TH, Levy C, Ersek M, Miller SC, Gidwani-Marszowski R, et al. Association of expanded VA hospice care with aggressive care and cost for veterans with advanced lung cancer. JAMA Oncol. 2019;5(6):810–6.
- 183. Haverhals LM, Manheim CE, Mor V, Ersek M, Kinosian B, Lorenz KA, et al. The experience of providing hospice care concurrent with cancer treatment in the VA. Support Care Cancer. 2019;27(4):1263–70.
- 184. Patient Protection and Affordable Care Act. In. Vol Pub. L. No. 111-148, 124 Stat. 119 §3140. 2010.
- 185. Abt Associates. Evaluation of the medicare care choices model. Annual report 2. Contract No. HHSM-500-2014-000261. Feb 2020. Task Order No. T0005. Rockville: Abt Associates; 2020. Available at: https://downloads.cms.gov/files/mccm-secannrpt. pdf. Accessed 25 Jul 2020.
- 186. Harrison KL, Connor SR. First Medicare demonstration of concurrent provision of curative and hospice services for end-of-life care. Am J Public Health. 2016;106(8):1405–8.
- 187. Mayer DK, Travers D, Wyss A, Leak A, Waller A. Why do patients with cancer visit emergency departments? Results of a 2008 population study in North Carolina. J Clin Oncol. 2011;29(19):2683–8.
- 188. Agency for Healthcare Research and Quality (AHRQ). Patient centered medical home resource center. Defining the PCMH. http://pcmh.ahrq.gov/page/defining-pcmh. Accessed 9 Sept 2020.
- 189. Scholle SH, Tirodkar M, Patel T, Smith-McLallen A, Friedberg M, Chanin J, et al. Evaluating a new patient-centered approach for cancer care in oncology offices. Washington: Patient-Centered Outcomes Research Institute (PCORI); 2020. https://doi.org/10.25302/1.2020.IH.121143.

- 190. Tirodkar MA, Acciavatti N, Roth LM, Stovall E, Nasso SF, Sprandio J, et al. Lessons from early implementation of a patient-centered care model in oncology. J Oncol Pract. 2015;11(6):456–61.
- 191. Centers for Medicare & Medicaid Services. Health care innovation awards round one project profiles. Last updated Dec 2013. http://innovation.cms.gov/Files/x/HCIA-Project-Profiles.pdf. Accessed 9 Sept 2020.
- 192. Waters TM, Webster JA, Stevens LA, Li T, Kaplan CM, Graetz I, McAneny BL. Community oncology medical homes: physiciandriven change to improve patient care and reduce costs. J Oncol Pract. 2015;11(6):462–7.
- 193. Waters TM, Kaplan CM, Graetz I, Price MM, Stevens LA, McAneny BL. Patient-centered medical homes in community oncology practices: changes in spending and care quality associated with the COME HOME Experience. J Oncol Prac. 2019;15(1):e56–64.
- 194. Thomas R, Wilson DM, Justice C, Birch S, Sheps S. A literature review of preferences for end-of-life care in developed countries by individuals with different cultural affiliations and ethnicity. J Hospice Palliat Nurs. 2008;10:142–63.
- 195. Weeks JC, Catalano PJ, Cronin A, Finkelman MD, Mack JW, Keating NL, Schrag D. Patients' expectations about effects of chemotherapy for advanced cancer. N Engl J Med. 2012;367(17):1616–25.
- 196. Chen AB, Cronin A, Weeks JC, Chrischilles EA, Malin J, Hayman JA, Schrag D. Expectations about the effectiveness of radiation therapy among patients with incurable lung cancer. J Clin Oncol. 2013;31(21):2730–5.
- 197. Pawlik TM, Devon KM, Fields CA, Hinshaw DB, Evidence-Based Reviews in Surgery Group. What are patients' expectations about the effects of chemotherapy for advanced cancer? J Am Coll Surg. 2014;219(3):588–90.
- 198. Hagerty RG, Butow PN, Ellis PM, Dimitry S, Tattersall MHN. Communicating prognosis in cancer care: a systematic review of the literature. Ann Oncol. 2005;16(7):1005–53.
- 199. Huskamp HA, Keating NL, Malin JL, Zaslavsky AM, Weeks JC, Earle CC, et al. Discussions with physicians about hospice among patients with metastatic lung cancer. Arch Intern Med. 2009;169(10):954–62.
- 200. Mack JW, Smith TJ. Reasons why physicians do not have discussions about poor prognosis, why it matters, and what can be improved. J Clin Oncol. 2012;30(22):2715–7.
- Medicare Prescription Drug, Improvement, and Modernization Act of 2003. In. Vol Pub. L. No. 108-173, 117 Stat. 2289. Sec. 501(b). 2003.
- 202. Centers for Medicare & Medicaid Services (CMS). Fiscal year 2009 quality measure reporting for 2010 payment update 2008. https://www.cms.gov/Medicare/Quality-Initiatives-Patient-Assessment-Instruments/HospitalQualityInits/Downloads/ HospitalRHQDAPU200808.pdf. Accessed 9 Sept 2020.
- 203. Jha AK, Li Z, Orav EJ, Epstein AM. Care in U.S. hospitals-the Hospital Quality Alliance program. N Engl J Med. 2005;353(3):265–74.
- 204. Deficit Reduction Act of 2005. In. Vol Pub. L. No. 109-171, 120 Stat. 28. Sec. 5001(a). 2006.
- 205. Tax Relief and Health Care Act of 2006. In. Vol Pub. L. No. 109-432, 120 Stat. 2984. Sec. 109(a). 2006.
- 206. Medicare Access and CHIP Reauthorization Act of 2015. In. Vol Pub. L. No. 114-10, 129 Stat. 87, §101. 2015.
- 207. National Quality Forum (NQF). National voluntary consensus standards for ambulatory care: specialty clinician performance measures: a consensus report. Washington: NQF; 2007. http:// www.qualityforum.org/WorkArea/linkit.aspx?LinkIdentifier=id& ItemID=60693. Accessed 9 Sept 2020.

- 208. Schuur JD, Hsia RY, Burstin H, Schull MJ, Pines JM. Quality measurement in the emergency department: past and future. Health Aff (Millwood). 2013;32(12):2129–38.
- 209. Graff L, Stevens C, Spaite D, Foody J. Measuring and improving quality in emergency medicine. Acad Emerg Med. 2002;9(11):1091–107.
- Welch S, Augustine J, Camargo CA, Reese C. Emergency department performance measures and benchmarking summit. Acad Emerg Med. 2006;13(10):1074–80.
- 211. Welch SJ, Asplin BR, Stone-Griffith S, Davidson SJ, Augustine J, Schuur J. Emergency Department Benchmarking Alliance. Emergency department operational metrics, measures and definitions: results of the Second Performance Measures and Benchmarking Summit. Ann Emerg Med. 2011;58(1):33–40.
- 212. Stone-Griffith S, Englebright JD, Cheung D, Korwek KM, Perlin JB. Data-driven process and operational improvement in the emergency department: the ED Dashboard and Reporting Application. J Healthc Manag. 2012;57(3):167–80.
- 213. National Quality Forum (NQF). National voluntary consensus standards for emergency care: a consensus report. Washington: NQF; 2009. http://www.qualityforum.org/WorkArea/linkit.aspx? LinkIdentifier=id&ItemID=22009. Accessed 9 Sept 2020.
- 214. National Quality Forum (NQF). Measurement framework: evaluating regionalized emergency medical care systems using an episodes of care approach. Washington: NQF; 2012. http://www. qualityforum.org/WorkArea/linkit.aspx?LinkIdentifier=id&Ite mID=70700. Accessed 28 Jul 2020.
- 215. National Quality Forum (NQF). Regionalized emergency medical care services: emergency department crowding and boarding, healthcare system preparedness and surge capacity – performance measurement gap analysis and topic prioritization: final report. Washington: NQF; 2012. http://www.qualityforum. org/WorkArea/linkit.aspx?LinkIdentifier=id&ItemID=72643. Accessed 28 Jul 2020.
- 216. National Quality Forum (NQF). Emergency department transitions of care: a quality measurement framework final report.

Washington: NQF; 2017. http://www.qualityforum.org/WorkArea/ linkit.aspx?LinkIdentifier=id&ItemID=85864. Accessed 28 Jul 2020.

- 217. National Quality Forum (NQF). Advancing chief complaint-based quality measurement: final report. Washington: NQF; 2019. http:// www.qualityforum.org/WorkArea/linkit.aspx?LinkIdentifier=id& ItemID=90454. Accessed 9 Sept 2020.
- 218. National Quality Forum (NQF). eMeasure trial use program. Washington: NQF; 2020. https://www.qualityforum.org/eMeasure_Trial_Use.aspx. Accessed 28 Jul 2020.
- 219. National Quality Forum (NQF). NQF time-limited endorsement. http://www.qualityforum.org/WorkArea/linkit.aspx?LinkIdentifie r=id&ItemID=20787. Accessed 9 Nov 2020.
- 220. National Quality Forum (NQF). Quality positioning system. Washington: NQF; 2015. http://www.qualityforum.org/Qps/ QpsTool.aspx. Accessed 9 Sept 2020.
- 221. Anderson KM, Marsh CA, Flemming AC, Isenstein H, Reynolds J. Quality measurement enabled by health IT: overview, possibilities, and challenges. (Prepared by Booz Allen Hamilton, under Contract No. HHSA290200900024I.) AHRQ Publication No. 12-0061-EF. Agency for Healthcare Research and Quality: Rockville; 2012.
- 222. Parsons A, McCullough C, Wang J, Shih S. Validity of electronic health record-derived quality measurement for performance monitoring. J Am Med Inform Assoc. 2012;19(4):604–9.
- 223. Kern LM, Kaushal R. Accuracy of electronically reported "meaningful use" clinical quality measures. Ann Intern Med. 2013;159(1):73.
- 224. Pronovost PJ, Lilford R. A road map for improving the performance of performance measures. Health Affair. 2011;30(4):569–73.
- 225. Spinks T, Ganz PA, Sledge GW Jr, Levit L, Hayman JA, Eberlein TJ, Feeley TW. Delivering high-quality cancer care: the critical role of quality measurement. Healthc (Amst). 2014;2(1):53–62.
- Lamb GC, Smith MA, Weeks WB, Queram C. Publicly reported quality-of-care measures influenced Wisconsin physician groups to improve performance. Health Affair. 2013;32(3):536–43.

Patient Navigation

David C. Seaberg and Donald Norris II

Introduction

Cancer patients face significant challenges. This is true not only with their illness but also when trying to navigate an increasingly matrix-oriented healthcare system. While advances in cancer treatment have helped save millions of lives over the last three decades, patients face far more complex treatment decisions and follow-up options than ever before. The amount of time required and the types of services cancer patients utilize have expanded across the spectrum of prevention, screening, diagnosis, treatment, and survivorship. As more cancer patients live with chronic illness, the length of time a patient is engaged with the healthcare system, including the emergency department (ED), has increased. Efforts made by healthcare systems to enhance patient experience and coordinate care have become increasingly more important.

Patient navigation is seen as one possible solution to this problem. Patient navigation has been shown to improve timely cancer care [1–4]. Patient navigation programs reduce gaps in care by improving access to cancer services, improving the timeliness of the provision of these services, and add-ing strong support and guidance to patients [5]. Patient navigation involves collaboration between patients, providers, families, and caregivers. Their collaboration extends throughout the cancer continuum, from prevention and screening through treatment, survivorship, and palliative and end-of-life care. These programs employ a variety of individuals from laypeople to trained nurses and social workers to assist patients.

US emergency departments are an increasingly important site of care for patients with complex care needs. Specifically, ED visits for patients with cancer now exceed 4.5 million annually [6, 7]. ED visits for individuals aged 65 years and older now number nearly 30 million [8]. Many of these patients lack essential health knowledge including an understanding of the resources they need to optimize their health. Patient navigator systems help patients with complex care needs navigate the multifaceted and fragmented medical systems.

Case Study

Charlotte, North Carolina-based Atrium Health's Levine Cancer Institute developed a patient navigation program that featured 25 nurse navigators spread across seven locations. Patients who did not receive navigation services were 52% more likely to have unplanned hospital readmissions within 30 days when compared with patients who did. In a retrospective cohort study of approximately 2300 patients with poor prognosis (as defined by the American College of Surgeons [ACS]), the program found a significant survival benefit for patients who had received navigation services compared with those who had not. The greatest survival advantage was seen in patients who were either African American, were insured by Medicaid, or had lung or pancreatic cancer [9].

Background

Patient navigation is an intervention to reduce health disparities in cancer care. It is specifically aimed at vulnerable or medically underserved populations. The first patient navigator program was created by Harold Freeman in 1990 at Harlem Hospital in New York, NY. He focused on underserved women with breast cancer [10, 11]. The goals of the program were to expand access to cancer screening, improve



D. C. Seaberg (🖂) · D. Norris II

Department of Emergency Medicine, Summa Health, Akron, OH, USA

Department of Emergency Medicine, Northeast Ohio Medical University, Rootstown, OH, USA e-mail: seabergd@summahealth.org

clinical follow-up among medically underserved women through community outreach, and reduce the time between an abnormal test result and diagnosis and/or treatment [12]. Eliminating barriers to health access, such as lack of insurance or cultural and communication barriers, was also critically important. The navigation program was remarkably successful with an increased 5-year survival of 31% [10, 11].

In the 30 years since Dr. Freeman's initial program, the use of patient navigator programs has spread to other cancer treatment programs as a means to decrease barriers to care and improve overall outcomes for disparate populations. Many of the current patient navigator programs were initially focused on providing guidance to the racial/ethnic minority or lowincome populations. In 2001, the President's Cancer Panel was established with the National Cancer Act of 1971. It revealed that barriers limiting or preventing access to cancer care are not unique to poor Americans but were experienced by Americans across all socioeconomic levels [13, 14]. The panel found multiple reasons why people of all socioeconomic levels were not getting screened and treated. These were primarily related to fear, literacy, culture, and religion [15, 16]. In 2005, the Outreach and Chronic Disease Prevention Act was signed, providing financial grants for the development and operation of patient navigator services to improve healthcare outcomes across the entire country. Each approved hospital was to add a full-time navigator program [17]. In 2012, the ACS Commission on Cancer released standards that reflected the goal of "ensuring patient-centered care." One of the new standards (Standard 3.1) required all cancer programs seeking accreditation to have a patient navigation program [18, 19].

As the importance of patient navigation has been recognized, the role of nurse or patient navigators and the terms used to describe these individuals and their functions within the oncologic care team have evolved.

The National Cancer Institute (NCI) describes patient navigation as the support and guidance provided to persons with abnormal screenings or new cancer diagnoses, including overcoming challenges and barriers to accessing the healthcare system in a culturally competent manner [20–23]. They state that the role of a navigator is to help patients and their families assess the cancer care system and overcome barriers to receiving care. This includes facilitating the provision of timely, quality care in a culturally sensitive manner [15, 24]. Specific time points in the cancer continuum are the focus, and the goal is to reduce the complexity of the healthcare system for the patient [25].

Management

Patient navigation is provided to individual patients for a defined episode of cancer-related care. The goal of patient navigation is to provide an intervention that addresses barri-

ers to quality standard care by providing individualized assistance to patients, survivors, and families [9, 26, 27]. Reducing time to access healthcare resources is critical for the patient. Oncologic navigators bridge the gap between health systems and the patient. An obvious byproduct of a successful cancer navigation program is decreased visits to the ED for these patients [28]. Navigators have the ability to effectively communicate with providers and other stakeholders within and across the institution while serving patients as knowledgeable, caring peers, and allies who have an inside track to the health system [29]. Patient navigators help patients address barriers ranging from communication to psychological or financial needs and health systems social support-related issues. Other common topics in oncologic navigation are [30]:

- · Financial and economic issues
- Differences in language, which may prevent patients and family from understanding treatment recommendations
- Cultural and ethnic diversity requiring tailored interventions
- Communication among healthcare teams, patients, families, and other healthcare providers
- Transportation problems that impact patients' ability to receive healthcare
- Emotional concerns for patients including distress and fear, which may prolong decision-making and delay interventions

Oncologic patient navigators perform many roles. The navigators help reduce barriers to healthcare services through educating patients on their diagnosis/treatment and build partnerships in the community. Additionally, they help coordinate appointments, maintain communication, connect patients and families with support services, teach self-advocacy, and provide access to clinical trials. There are multiple roles in patient navigation yet no universally accepted definition. There is no consensus on necessary preparation and competencies for filling the role [31–33].

Oncologic Nurse Navigator

An oncologic nurse navigator (ONN) is a professional registered nurse with oncologic-specific clinical knowledge who offers individualized assistance to patients, families, and caregivers to help overcome healthcare system barriers and facilitate informed decisions [31, 32]. They ensure that the patient receives timely access to quality health and psychosocial care throughout all phases of the cancer continuum. In 2009, the National Coalition of Oncologic Nurse Navigators (NCONN) developed the first competencies that defined the role of the ONN [34–36]. Developed in consultation with active professional oncologic nurse navigators throughout the United States, these core competencies cover five areas: (1) professional, legal, and ethical nursing practice; (2) health promotion and health education; (3) management and leadership; (4) negotiating the healthcare delivery system and advocacy; and (5) personal effectiveness and professional development [36, 37]. The first published guidelines establishing core competencies for ONNs were established by the Oncologic Nursing Society in 2013 [37, 38].

Oncologic Social Worker

Oncologic social workers are knowledgeable about cancer and the psychosocial and other effects of disease, treatment, and survivorship. They have advanced degrees and have additional experience and training in oncologic and other life-threatening illnesses. The oncologic social work navigator helps foster coping and adaption to cancer and its effects in order to help cancer survivors maintain or improve quality of life [38]. Their focus in clinical practice is to help complete a psychosocial assessment to determine survivor and family strengths and needs relative to coping effectively with cancer diagnosis, treatment, and follow-up care [39]. They work with other team members to develop a multidisciplinary care plan and help provide case management services. They also provide direct assistance to help meet financial, transportation, lodging, and other psychosocial needs, including end-of-life advance care planning [39]. Oncologic social workers also assist survivors in navigating through healthcare systems to help them achieve quality care. They help provide social and emotional support to cancer survivors by mobilizing new or existing family, system, and community resources [30]. In addition, oncologic social work navigators advocate with, or on behalf of, survivors, families, and caregivers to address their needs as well as advocate for policies and programs that will benefit them.

Lay Navigator

A lay navigator is a trained nonprofessional or volunteer who provides individualized assistance to patients, families, and caregivers [40]. They help overcome healthcare system barriers and facilitate timely access to quality healthcare and psychosocial care, from pre-diagnosis through all phases of cancer care. Interestingly, Dr. Freeman's original patient navigation program believed that the role of the navigator should be served by a layperson and not by a nurse or social worker [41].

Navigators can be disease, condition, or service specific, with many specializing in areas such as breast, prostate, or lung cancer. Some navigators focus primarily on the role of financial navigator. Others may provide assistance in the areas of screening and testing, as the original navigators did. Patient navigation in cancer care currently focuses on assisting patients in the coordination of care among providers, the community, and patients and their families [12, 42–44]. To move patient navigation forward, the cancer community will need to develop and agree upon a streamlined definition that can be promoted by all stakeholders and understood by patients and their families [45].

Patient-Centered Medical Home

The concept of a patient-centered medical home (PCMH) originated in pediatric literature dating to the late 1960s. Originally conceived as a way to integrate care for children with special needs, physicians soon realized that fragmented care was seen in the general pediatrics population as well. In 2007, the American College of Physicians, American Academy of Family Physicians, American Osteopathic Association, and American Academy of Pediatrics drafted the Joint Principles of the Patient-Centered Medical Home. This document defined the PCMH placing the relationship between the patient and their primary care physician at the center. While the primary physician maintains a longitudinal relationship with the patient, they also lead the team of other individuals integral to the overall care of the patient. Emphasis was placed on quality and safety leading to an overall reduction in cost with higher quality of care [46].

An additional positive outcome often mentioned when discussing the PCMH is a reduction in visits to the emergency department. The American College of Emergency Physicians (ACEP) released a policy on patient-centered medical homes in 2008 revising the policy in 2015. In this policy, ACEP fully supports the PCMH if it upholds the following principles:

- 1. Provide high-quality, safe, and efficient medical care
- Provide patient access to a personal physician, the leader of a team of individuals who collectively take responsibility for ongoing care of their patients
- Ensure patients have the freedom to select a specialist of their choosing and access emergency medical care when they feel they need it
- 4. Include the safety net of emergency care [47]

In 2012, the Michigan Oncologic Medical Home Demonstration Project was developed. It was one of the first multi-practice oncologic medical home projects in the country. Placing the oncologist and patient at the center of the design, this group standardized development of treatment regimens including deployment of enhanced triage and access protocols. The group also standardized advance care planning across the groups. In addition, a medical home nurse line was established to help triage patients seeking unscheduled care. In the first year of the 85-patient pilot program, the overall cost savings was \$550 per patient [48].

Health Services/Resource Utilization

Patient navigation programs and the backgrounds of those who serve as patient navigators are driven by local needs. There is no one type of patient navigation model that fits the needs of all medical settings or systems. Healthcare organizations that start patient navigation programs need to assess the needs of their populations and tailor the intervention to those needs. Patient navigation is typically a goal-oriented intervention that focuses on reducing the barriers to achieving a particular cancer healthcare goal, such as improvements in cancer screening rates, cancer treatment adherence, or patient satisfaction with care [12, 23]. Particular emphasis on implementing patient navigation interventions should focus on improvements in a particular outcome of interest. Although individual patients may benefit from actions taken by patient navigators across the cancer care continuum, current literature shows that interventions that target cancer screening outcomes have the greatest clinical benefit and most cost-effectiveness [49].

Patient navigation programs have had many successes. The Patient Navigation Research Program (PNRP) was designed to develop interventions to reduce the time to diagnosis and treatment of cancer after identifying an abnormal finding from a cancer detection procedure. The PNRP studies, enrolling over 10,000 patients, found that patient navigation reduced the time from abnormal findings to diagnosis in breast, cervical, colorectal, and prostate cancers [50]. Resolutions of cancer therapy at 180 days and at 270 days with navigation were 56.2% and 70.0%, respectively, compared with 53.8% and 68.2%, respectively, with usual care. The estimated cost of navigation in this study was \$275 per patient.

Patient navigation services have proven to be effective at reducing the cost and improving the quality of care for patients with cancer. Patient navigation has been found in survey studies to increase patient and staff satisfaction while decreasing barriers to care. One study published by The Ralph Lauren Center for Cancer Care and Prevention in New York found improved cost-effectiveness for patients with breast and colorectal cancer [51]. Researchers found that patient navigation led to patients achieving diagnostic resolution with shorter treatment intervals. There were significant cost savings per patient at the end of treatment. A study at the George Washington Cancer Institute in Washington, D.C., found that the patient navigation program significantly reduced breast cancer diagnostic time. Use of patient navigation in the emergency department was found to decrease overall patient and hospital costs as well as reduce subsequent emergency department visits [4].

Adding care coordination to patient navigation has proven to have many benefits to physicians, patients, and their families. It enhances interaction among physicians, increases patient awareness and self-management, and improves patient education and satisfaction [52]. It also increases referrals for nutrition and physiological care as well as physical and social service needs [52, 53]. For the patient and the family, care coordination simplifies the process. Care coordination also facilitates access to ancillary and community services and resources. Other benefits include improved education of patient and family, increased collaboration and communication among the patient and the healthcare team, improved patient self-care and satisfaction, and increased multidisciplinary care [23].

ED visits have increased exponentially over the last three decades now topping 140 million per year [8]. Long recognized as a safety net provider, the ED is a usual source of care for those who are affected by barriers to continuity relationships with physicians as well as patients with complex medical problems requiring a coordinated care approach. Initial evidence suggests that patient navigators may help remove barriers to timely, effective patient care among vulnerable populations which may, in turn, influence ED utilization [54, 55]. In a survey of oncologic patient navigators, 91% indicated that lay navigation services would be either very helpful or moderately helpful for ED patients with cancer. Coordination of care, the provision of emotional support and educational resources, and providing companionship to older persons during the ED visits were identified as priorities for an ED-based lay navigation program. The three most common barriers that emergency department patient navigators help patients with include insurance and out-of-pocket expenses, transportation issues, and helping manage the feelings and fears associated with cancer [56]. Follow-up appointments and getting prescriptions filled were also important considerations. Funding for ED navigation programs came initially through small foundational grants for pilot programs. As the value and role of navigation programs have become more defined, hospitals have increasingly funded ED navigators who consist of teams of transitional care coordinators, case managers, and lay personnel.

There have been numerous emergency department studies on the use of patient navigation in patients who utilize the ED frequently. It has been reported that just 5% of ED patients comprise nearly 25% of all ED visits yearly [57, 58]. Though the reasons for this are multifactorial including lowincome status, lack of insurance, severe chronic illnesses, and lack of primary care provider and familiarity, the cost to the patient and the community is exceedingly high [59, 60]. The cost to both the patient and community is not just monetary; care of these patients is inappropriately fragmented. ED patients who frequently utilize the ED contribute to overcrowding affecting the community's emergency readiness. In one study, paid social work staff were trained to help patients navigate their healthcare options during their ED visit by educating the patient on their diagnosis, reviewing the associated treatment and follow-up plans including medications, and identifying proper primary care services and/or community resources that can aid with the patient's specific issues [58]. This study found that a patient navigator program reduced 1-year ED visits and costs by 9%. It was found that the majority of high utilizers had primary care providers as well as insurance. Lack of health literacy impacts their ability to navigate the health system. Lack of communication, lack of preventative care, referral options, and a cultural distrust led to poor outcomes and overutilization of the ED. Patient navigation has been shown to improve health outcomes and reduce costs [61, 62].

Another study by the Highmark Foundation found that patient navigators in three health systems were able to reduce excessive emergency department visits by 43% over 1 year [28]. In addition to the decrease in emergency department visits, one health system reported a 60% reduction in 30-day readmission, and another reported increased colonoscopy screening by 13%. The certified patient navigators in the study were laypersons who helped with nonclinical tasks such as picking up prescriptions, arranging transportation, connecting patients to community resources, and conducting post-discharge follow-ups [28].

A study at the Memorial Hermann Health System in Houston developed a patient navigation program designed to promote appropriate primary and oncologic care utilization and prevent or reduce ED use [63]. The navigators consisted of bilingual, state-certified community health workers who were trained in peer-to-peer counseling and connected medically underserved patients with medical homes and related support services. The navigators provided education about the importance of follow-up care, assisted with appointment scheduling, and followed up with patients to monitor and address additional care barriers. Over a 12-month period, ED visits declined as did medical costs [63].

Future Needs/Vision

Patient navigation needs to define the roles within the patient navigation team. Research has shown that there are similarities between the improvements in outcomes associated with patient navigation and those associated with care coordination [24]. One study reported that coordination and continuity of care were similarly important to the roles of patient navigator, nurse care manager or coordinator, and nurse navigator, for improving outcomes [33]. Another report suggested that professionally led models of patient navigation

that were similar to comprehensive case management models—adding the supportive functions of advocacy, education, problem solving, and support to the standard model of assessment, outreach, and referrals—were favorable to other models [33, 64]. Coordination of care and ensuring continuity of care were described as overarching roles and responsibilities in an integrative review [64]. The future of patient navigation revolves around integrating the navigation team within the patient-centered medical home. Defining the roles within the navigation team will be essential to achieving specific clinical outcomes as well as creating efficiency and cost-effectiveness. In order to achieve their effectiveness, patient navigation programs will need to expand more to outpatient areas including the emergency department.

The new standard set by the ACS CoC in establishing patient navigation programs in all ACS CoC-accredited facilities has led to enhanced funding for navigation programs [64]. Further work is needed to demonstrate the value of the blended model patient navigation care coordination. Further studies will be needed to evaluate the economic impact of the patient navigation programs. Cost-effectiveness analysis and budget impact studies are needed to further refine these patient navigation programs in the patientcentered medical home. The Health Services Research (HSR) cost workgroup of the American Cancer Society National Patient Navigator Leadership Summit met to examine cost data relevant to assessing the economic impact of patient navigation and to propose common cost metrics. Five categories of core and optional cost measures were identified: program costs, human capital costs, direct medical costs, direct nonmedical costs, and indirect costs [64]. The researchers recommended adoption of these metrics to promote understanding of the economic impact of patient navigation and comparability across diverse patient navigation programs. Further funding opportunities will need to be explored to enhance the development of these patient navigator programs.

Conclusion

Models of patient navigation and care coordination must continue to evolve to meet the ever-changing needs of cancer patients and their families. New models must hold fast to the guiding principles of the original navigation program: improve access to the cancer care system and facilitate quality care. These programs will not only lead to more rapid diagnoses, improved patient outcomes, and increased cost savings, but they will also ensure greater patient adherence to treatment programs. With the increasing use of clinical pathways in oncologic, and the need to demonstrate the value of these pathways, navigation and care coordination programs can be effective tools for maximizing the quality of care provided to patients. Expansion into the outpatient setting and emergency department may further increase the effectiveness and lead to both improved clinical outcomes and cost savings.

References

- Crane-Okada R. Evaluation and outcome measures in patient navigation. Semin Oncol Nurs. 2013;29(2):128–40.
- 2. Paskett ED, Harrop JP, Wells KJ. Patient navigation: an update on the state of the science. CA Cancer J Clin. 2011;61:237–49.
- National Cancer Act of 1971, HR 531, 112th Cong, 2nd Sess. 1971. https://smokershistory.com/NCA1971.htm. Accessed 3 Apr 2020.
- Hoffman HJ, LaVerda NL, Young HA, Levine PH, Alexander LM, Brem R, et al. Patient navigation significantly reduces delays in breast cancer diagnosis in the District of Columbia. Cancer Epidemiol Biomark Prev. 2012;21(10):1655–63.
- Wells KJ, Valverde P, Ustjanauskas AE, Calhoun EA, Risendal BC. What are patient navigators doing, for whom, and where? A national survey evaluating the types of services provided by patient navigators. Patient Educ Couns. 2018;101(2):285–94.
- Mayer DK, Travers D, Wyss A, Leak A, Waller A. Why do patients with cancer visit emergency departments? Results of a 2008 population study in North Carolina. J Clin Oncol. 2011;29(19):2683–8.
- Siefert ML, Bonquist TM, Berry DL, Hong F. Symptom-related emergency department visits and hospital admissions during ambulatory cancer treatment. J Community Support Oncol. 2015;13(5):188–94.
- Rui P, Kang K. National Hospital Ambulatory Medical Care Survey: 2017 emergency department summary tables. National Center for Health Statistics. Available from: https://www.cdc.gov/nchs/data/ nhamcs/web_tables/2017_ed_web_tables-508.pdf. Accessed 3 Apr 2020.
- Kline RM, Rocque GB, Rohan EA, Blackley KA, Cantril CA, Pratt-Chapman ML, et al. Patient navigation in cancer: the business case to support clinical needs. J Oncol Pract. 2019;15(11):585–90.
- Freeman HP, Muth BJ, Kerner JF. Expanding access to cancer screening and clinical follow-up among the medically underserved. Cancer Pract. 1995;3(1):19–30.
- Freeman HP. The origin, evolution, and principles of patient navigation. Cancer Epidemiol Biomark Prev. 2012;21(10):1614–7.
- Shockney LD. Evolution of patient navigation. Clin J Oncol Nurs. 2010;14(4):405–7.
- Freeman H, Reuben S. Voices of a broken system: real people, real problems. President's Cancer Panel 2000–2001 annual report. Bethesda: NCI; 2002. http://deainfo.nci.nih.gov/advisory/pcp/ archive/pcp00-01rpt/PCPvideo/voices_files/index.html. Accessed 20 Jan 2016.
- American Cancer Society. The history of cancer. 2014. Last medical review: 12 Jun 2014. Last revised: 4 Jan 2018. https://www. cancer.org/cancer/cancer-basics/history-of-cancer.html. Accessed 19 Dec 2019.
- Riley S, Riley CR. The role of patient navigation in improving the value of oncology care. J Clin Pathw. 2016;2(1):41–7.
- Freund KM, Battaglia TA, Calhoun E, et al. National Cancer Institute Patient Navigation Research Program: methods, protocol, and measures. Cancer. 2008;113(12):3391–9.
- H.R. 1812 109th Congress: Patient Navigator Outreach and Chronic Disease Prevention Act of 2005. www.GovTrack.us. 2005. https://www.govtrack.us/congress/bills/109/hr1812. Accessed 19 Dec 2019.
- American College of Surgeons: Commission on Cancer. Cancer Program Standards 2012: Ensuring patient-centered care. V.1.2.1.

Chicago: American College of Surgeons; 2012. https://www.facs. org/~/media/files/quality%20programs/cancer/coc/programstandards2012.ashx%20ACS%20CoC;%202012. Accessed 20 Jan 2019.

- Hede K. Agencies look to patient navigators to reduce cancer care disparities. J Natl Cancer Inst. 2006;98(3):157–9.
- Epstein RM, Street RL Jr. Patient-centered communication in cancer care: promoting healing and reducing suffering, NIH Publication No. 07-6225. Bethesda: National Cancer Institute; 2007.
- Newman-Horm PA. C-Change. Cancer patient navigation. Washington, DC: C-Change; 2005.
- Corporate Research Associates Inc. Cancer patient navigation: evaluation findings. (CCNS 461-7005Q). Halifax: Cancer Care Nova Scotia; 2004.
- 23. Jean-Pierre P, Hendren S, Fiscella K, Loader S, Rousseau S, Schwartzbauer B, et al. Understanding the processes of patient navigation to reduce disparities in cancer care: perspectives of trained navigators from the field. J Cancer Educ. 2011;26(1):111–20.
- Cantril CA. Overview of nurse navigation. In: Blaseg KD, Daugherty P, Gamblin KA, editors. Oncology nurse navigation: delivering patient-centered care across the continuum. Pittsburgh: Oncology Nursing Society; 2014. p. 1–13.
- 25. Freund KM, Battaglia TA, Calhoun E, Dudley DJ, Fiscella K, Paskett E, Patient Navigation Research Program Group, et al. National Cancer Institute Patient Navigation Research Program: methods, protocol, and measures. Cancer. 2008;113(12):3391–9.
- 26. Wells KJ, Battaglia TA, Dudley DJ, et al. Patient navigation: state of the art, or is it science? Cancer. 2018;113:1999–2010.
- Pedersen A, Hack TF. Pilots of oncology health care: a concept analysis of the patient navigator role. Oncol Nurs Forum. 2010;37(1):55–60.
- Accenture. Insight driven health. Patient navigation eases clinical workforce challenges – without clinical expertise. 2014. https:// www.accenture.com/t20150523T022442_w_/us-en/_acnmedia/Accenture/Conversion-Assets/DotCom/Documents/Global/ PDF/Dualpub_1/Accenture-Patient-Navigation-Eases-Clinical-Workforce-Challenges.pdf#zoom=50. Accessed 3 Apr 2020.
- 29. Freeman HP. The history, principles, and future of patient navigation. Semin Oncol Nurs. 2013;29(3):72–5.
- 30. Oncology Nursing Society; Association of Oncology Social Work; National Association of Social Workers. Oncology Nursing Society, the Association of Oncology Social Work, and the National Association of Social Workers joint position on the role of oncology nursing and oncology social work in patient navigation. Oncol Nurs Forum. 2010;37(3):251–2.
- Cantril C, Haylock PJ. Patient navigation in the oncology care setting. Semin Oncol Nurs. 2013;29(2):76–90.
- Academy of Oncology Nurse and Patient Navigators. Helpful definitions. Navigation process. http://www.aonnonline.org/about/definitions. Accessed 15 Feb 2020.
- Brown CG, Cantril C, McMullen L, Barkley DL, Dietz M, Murphy CM, Fabrey LJ. Oncology nurse navigator role delineation study: an Oncology Nursing Society report. Clin J Oncol Nurs. 2012;16(6):581–5.
- DeGroff A, Coa K, Morrissey KG, Rohan E, Slotman B. Key considerations in designing a patient navigation program for colorectal cancer screening. Health Promot Pract. 2014;15(4):483–95.
- Krok-Schoen JL, Oliveri JM, Paskett ED. Cancer care delivery and women's health: the role of patient navigation. Front Oncol. 2016;6:2.
- Pratt-Chapman M, Willis A, Masselink L. Core competencies for oncology patient navigators. J Oncol Navig Surviv. 2015;6:16–21.
- 37. Cook S, Fillion L, Fitch M, Veillette AM, Matheson T, Aubin M, et al. Core areas of practice and associated competencies for nurses working as professional cancer navigators. Can Oncol Nurs J. 2013;23(1):44–62.

- Scope of Practice. Oncology social work toolbox. Scope of practice in oncology social work. 2001. https://www.aosw.org/professionaldevelopment/scope-of-practice/. Accessed 19 Jan 2020.
- Willis A, Reed E, Pratt-Chapman M, Kapp H, Hatcher E, Vaitones V, Collins S, et al. Development of a framework for patient navigation: delineating roles across navigator types. J Oncol Navig Surviv. 2013;4(6):20–6.
- 40. Rocque GB, Pisu M, Jackson BE, Kvale EA, Demark-Wahnefried W, Martin MY, Patient Care Connect Group, et al. Resource use and medicare costs during lay navigation for geriatric patients with cancer. JAMA Oncol. 2017;3(6):817–25.
- Freeman HP. A model patient navigator program. Oncol Issues. 2004;19:44–6.
- 42. Page RD, Newcomer LN, Sprandio JD, McAneny BL. The patientcentered medical home in oncology: from concept to reality. Am Soc Clin Oncol Educ Book. 2015:e82–9. https://doi.org/10.14694/ EdBook_AM.2015.35.e82. PMID: 25993243.
- 43. Rocque GB, Partridge EE, Pisu M, Martin MY, Demark-Wahnefried W, Acemgil A, et al. The Patient Care Connect Program: transforming health care through lay navigation. J Oncol Pract. 2016;12(6):e633–42.
- 44. Hedlund N, Risendal BC, Pauls H, Valverde PA, Whitley E, Esparza A, et al. Dissemination of patient navigation programs across the United States. J Public Health Manag Pract. 2014;20(4):E15–24.
- 45. Domingo JB, Davis EL, Allison AL, Braun KL. Cancer patient navigation case studies in Hawai'i: the complimentary role of clinical and community navigators. Hawaii Med J. 2011;70(12):257–61.
- O'Dell ML. What is a patient-centered medical home? Mo Med. 2016;113(4):301–4.
- Patient-centered medical home model (editorial). Ann Emerg Med. 2016;68(1):147–8.
- Kuntz G, Tozer JM, Snegosky J, Fox J, Neumann K. Michigan Oncology Medical Home Demonstration Project: first-year results. J Oncol Pract. 2014;10(5):294–7.
- Swan BA, Conway-Phillips R, Griffin KF. Demonstrating the value of the RN in ambulatory care. Nurs Econ. 2006;24(6):315–22.
- 50. Bensink ME, Ramsey SD, Battaglia T, Fiscella K, Hurd TC, McKoy JM, Patient Navigation Research Program, et al. Costs and outcomes evaluation of patient navigation after abnormal cancer screening: evidence from the Patient Navigation Research Program. Cancer. 2014;120(4):570–8.
- Donaldson EA, Holtgrave DR, Duffin RA, Feltner F, Funderburk W, Freeman HP. Patient navigation for breast and colorectal cancer in 3 community hospital settings: an economic evaluation. Cancer. 2012;118(19):4851–9.
- 52. Francz S. Today's nurse navigator: educating and advocating for cancer patients. Oncology Nursing News. 9 Nov 2012. https://

www.oncnursingnews.com/publications/oncology-nurse/2012/ october-2012/todays-nurse-navigator-educating-and-advocatingfor-cancer-patients. Accessed 19 Jan 2020.

- Oncology Nursing Society. Oncology nurse navigator core competencies. Pittsburgh: Oncology Nursing Society; 2013. https://www. ons.org/sites/default/files/ONNCompetencies_rev.pdf. Accessed 19 Jan 2020.
- Rivera DR, Brown J, Shelburne N. Questions about emergency department treatment of patients with solid tumors and hematological malignant neoplasms-reply. JAMA Oncol. 2018;4(5):750–1.
- 55. Carter N, Valaitis RK, Lam A, Feather J, Nicholl J, Cleghorn L. Navigation delivery models and roles of navigators in primary care: a scoping literature review. BMC Health Serv Res. 2018;18(1):96.
- 56. Bischof JJ, Sellers JB, Phillips AW, Petrongelli JJ, Stuckey AE, Platts-Mills TF. Patient navigation for complex care patients in the emergency department: a survey of oncology patient navigators. Support Care Cancer. 2019;27(11):4359–62.
- Shumway M, Boccellari A, O'Brien K, Okin RL. Cost-effectiveness of clinical case management for ED frequent users: results of a randomized trial. Am J Emerg Med. 2008;26(2):155–64.
- Seaberg D, Elseroad S, Dumas M, Mendiratta S, Whittle J, Hyatte C, Keys J. Patient navigation for patients frequently visiting the emergency department: a randomized, controlled trial. Acad Emerg Med. 2017;24(11):1327–33.
- 59. Roby DH, Pourat N, Pirritano MJ, Vrungos SM, Dajee H, Castillo D, Kominski GF. Impact of patient-centered medical home assignment on emergency room visits among uninsured patients in a county health system. Med Care Res Rev. 2010;67(4):412–30.
- Sarver J. Usual source of care and nonurgent emergency department use. Acad Emerg Med. 2002;9(9):916–23.
- Whitley E, Valverde P, Wells K, Williams L, Teschner T, Shih YC. Establishing common cost measures to evaluate the economic value of patient navigation programs. Cancer. 2011;117(15 Suppl):3618–25.
- 62. Kowalkowski MA, Raghavan D, Blackley K, Morris V, Farhangfar C. Patient navigation associated with decreased 30-day all-cause readmission. ASPO 40th Annual Meeting Abstracts. Cancer Epidemiol Biomarkers Prev. 2016;25(3):558. https://cebp.aacrjournals.org/content/25/3/558.2
- Enard K. Reducing preventable emergency department utilization and costs by using community health workers as patient navigators. J Health Manag. 2013;58(6):412–27.
- Institute of Medicine. The future of nursing: leading change, advancing health. Washington, DC: The National Academies Press; 2011. https://doi.org/10.17226/12956. Accessed 19 Aug 2020.



Petra Grami, Sorayah S. Bourenane, Danielle Milling, Karen McFarland, Robert T. Drew, and Linda Koppy

Emergency Nursing and Oncologic Emergencies

In 2011, the American Nurses Association recognized emergency nursing as a specialty practice. As a challenging and unique profession, this clinical practice area prepares nurses to provide prompt interventions to stabilize or prevent further patient deterioration. The fast-paced, high-acuity setting commands refined critical thinking, clinical assessment, communication, and prioritization skills. Additionally, emergency nurses (ENs) care for a diverse patient population with various medical conditions and acute care needs [1].

Of significance is the Emergency Nursing Association (ENA), founded in 1970 as the premier professional nursing association, devoted to illustrating the future of emergency nursing. The mission of ENA is to further excellence in emergency nursing [2].

Reflective of the diverse nature of healthcare, emergency medicine has multiple subspecialties. As such, different emergency departments (EDs) provide care to specific patient populations and health concerns. Some of the subtypes include trauma, stroke, cardiac, burn, neuro, disaster response, pediatric, and adult [3].

The same variations that exist in emergency medicine and EDs are also depicted among ENs. It is not surprising to find that most ENs assume multiple roles over the course of their career. Some of these roles include *trauma*, *triage*, *flight*, *critical care transport*, *pediatric, burn center, geriatric, and charge nurse* [3].

Professional advancement has also led to further specialization of ENs. This year, the Board of Certified Emergency Nursing recognized four different subspecialty professional certifications, including the *Certified Emergency Nurse*, *Certified Flight RN, Certified Pediatric Emergency Nurse*, *Certified Transport Registered Nurse, and Trauma Certified Registered Nurse* [3]. Notably, as professional advancement continues to detect emerging areas of subspecialty in emergency nursing, additional professional certifications, education, and targeted programs have been created to support quality care [3].

One emerging subspecialty is Oncologic Emergency Nursing (OEN), a clinical practice designed to provide care for patients presenting with cancer emergencies.

Consider this: the 5-year survival rate for all cancers increased from 39% in 1960 to nearly 70% in 2019 and thanks in part to new cancer treatment modalities, increased survival rates have led to approximately 15 million people in the United States with cancer. The majority of these individuals likely experience oncologic emergencies requiring an ED visit [4]. OEN is among the healthcare teams that assume multiple roles in the ED to provide quality care and ensure optimum health outcomes for the patients presenting with oncologic emergencies.

This chapter identifies high-frequency and high-risk conditions that an OEN may encounter when caring for an ED oncologic patient. For each condition, we then present a number of case studies to illustrate several medical diagnoses, associated risk factors, presenting signs and symptoms, potential causes, contributing factors, anticipated diagnostics, corresponding interventions, case dispositions, and specialty nursing considerations. We conclude with discussions of the impact of the pandemic (novel coronavirus) on cancer patients and an overview of regulatory bodies that guide ED practice in the oncologic setting.

Nursing

P. Grami (🖂)

Critical Care, Clinical Decision Unit and Acute Dialysis, Nursing Administration, MD Anderson Cancer Center, Houston, TX, USA e-mail: pgrami@mdanderson.org

S. S. Bourenane $(\boxtimes) \cdot D.$ Milling $\ \cdot \ K.$ McFarland $\cdot \ R.$ T. Drew L. Koppy

Emergency Department and Clinical Decision Unit, The University of Texas MD Anderson Cancer Center, Houston, TX, USA e-mail: sbourenane@mdanderson.org

K. H. Todd et al. (eds.), Oncologic Emergency Medicine, https://doi.org/10.1007/978-3-030-67123-5_6

Triage and General Assessment of Oncologic Emergencies

Triage

At triage, patients presenting to ED will encounter the triage nurse as the first medical professional. The role of the triage nurse is to identify patients needing immediate attention and then prioritize among those who do not require immediate life-saving interventions. Triage tools, such as the Emergency Severity Index (ESI), are used to standardize the approach and to predict patient disposition and ED resource use [5]. The ESI is a reliable and valid tool that is now in its fourth version [5], and is the most commonly used ED triage system in the United States [6]. The tool provides clinically relevant grading of patients into five groups from 1 (*most urgent*) to 5 (*least urgent*). This stratification is based on patients' acuity and resource needs [7].

Typically, patients with cancer have an ESI level of 1, 2, or 3 because of their potential to use multiple resources due to the complexity of their condition. Beyond the primary assessment is the need to screen for patients' cancer diagnoses, last treatment and modality, and any other significant surgeries or procedures [7].

General Assessment

ED nurses' ability to perform an accurate initial comprehensive patient assessment after triage is imperative in order to recognize the urgency and treatment needs of patients and to develop baseline data from which any changes in the condition of patients may be measured against [8].

Omission of accurate and timely patient assessments has been reported to result in adverse patient outcomes [9].

The EN can also expect the medical team to order laboratory tests. General labs should be assessed on almost all cancer patients, as treatment and disease processes may alter values from day to day and can likely provide additional insight into the underlying issue. Obtaining a complete blood count (CBC) with differential and comprehensive metabolic panel (CMP) is indicated for the vast majority of patients with a concurrent cancer diagnosis. Determining a patient's pancytopenia status, electrolyte levels and overall metabolic status are critical, as many abnormalities may not have associated symptoms in early stages. Early detection and correction of imbalances may prevent patients from incurring further injury or deterioration [10].

Although patients with cancer may present with an array of medical emergencies, this population is also at a high risk for infection. As such, sepsis should always be considered. Any signs or symptoms of infection should be promptly noted and addressed, as these patients are immunocompromised and have little physiological defense mechanisms. The presence of a central line also increases infection risk [11]. Timely collection of blood cultures and prompt administration of broad-spectrum antibiotics for any suspected infection can significantly improve the patient's prognosis [12].

Frequent ED encounters for patients with advanced cancer may also indicate a patient nearing end of life [13]. Patients with advanced cancer may benefit from conversations about advance care planning and code status to ensure their wishes are followed in the care trajectory. Although such conversations may be difficult, they can significantly improve the patient and family experience. Sending patients home with hospice from the ED is, at times, a feasible and appropriate option [14].

Chief Complaint: Chest Pain and Shortness of Breath

Patients presenting with chest pain and shortness of breath should be evaluated immediately for acute myocardial infarction via EKG. If EKG results do not suggest an acute MI, additional diagnostics should be employed to determine the underlying issue [15]. Chest pain and shortness of breath in cancer patients can be caused by various conditions that are common among cancer patients. These include pulmonary embolism, pleural effusion, pneumonia, spontaneous pneumothorax due to tumor burden, pericardial effusion, and cardiac tamponade [16]. See case studies in Table 6.1 illustrating a number of presentations involving these chief complaints [17–25]. Due to coagulopathies and bleeding tendencies, cancer patients may be at higher risk of cardiopulmonary-related adverse events [26]. As increasing numbers of patients receive immune checkpoint therapy, some cardiac presentation may represent immune-related adverse effects [27].

Tumors are known to cause collections of fluid proximal to tumor location. Depending on known lesions, tumors in the thoracic cavity may provide further insight into contributing conditions [28]. Pleural effusions are a high-frequency finding in lung cancer patients or patients with lung metastases. In recurrent pleural effusions, patients may have a Denver catheter drain placed to manage the fluid collection and reduce symptoms. If pleural effusion is identified, patients may require thoracentesis. This will likely alleviate any shortness of breath or chest pain symptoms almost immediately [29].

Table 6.1 Case studies: chest pain and shortness of breath

		Risk factors/contributing	Potential tests/		
Vital signs	Signs and symptoms	factors	interventions	Nursing considerations	
Pulmonary embolism. A 37-year-old female with myosarcoma currently on treatment presents with sharp left-sided chest pain and SOB since 0800 today. She has a past medical history of right deep vein thrombosis. The pain worsened with motion and deep breathing. The pain has been progressively increasing in severity and she now has severe left back and shoulder pain. She complains of SOB and "feels like she is going to die." She denies cough, fever, sputum production, or hemoptysis					
BP 121/84 HR 121 T 37.2 R 25 SpO ₂ 89%	Symptoms and clinical presentation may vary depending on the size of embolus & preexisting cardiopulmonary status, including asymptomatic/ incidental findings via outpatient diagnostic imaging studies Chest pain, often angina type onset and worsens with deep breathing. Chest pain will progress to pleuritic Dyspnea with sudden onset, tachypnea, crackles/wheezes, diminished breath sounds Tachycardia Low-grade fever (~40% of patients with PE) Cough/hemoptysis Syncope Back or abdominal pain Diaphoresis	Venous stasis Coagulopathies (may be induced by therapy or disease process) Atrial fibrillation Increased fatigue/ decreased activity Lung disease	General labs ECG Oxygen support CT with contrast to identify PE Arterial blood gases D-dimer (not as specific in cancer population) Initiation of thrombolytic therapy	Teach self-injection of low molecular weight heparin Monitor cardiopulmonary status for changes/ deterioration, may be rapid onset IV catheter = 18 g minimum for chest CT PE study protocol	

Pleural effusion. A 68-year-old man with lung cancer, congestive heart failure, and a 40-year history of cigarette smoking two packs a day. The patient reports that he has stopped smoking because of the lung cancer diagnosis. The patient presents to ED with shortness of breath and right-sided chest pain that worsens with deep breathing.

0	1 0			
BP 150/70	Signs/symptoms dependent on	CHF	General labs	Information about
HR 104	amount and rate of fluid	Pneumonia	Chest X-ray and/or CT	procedure
Т 36.6	accumulation	Malignancy	scan	Treatment plan and
R 26	Dyspnea	Pulmonary embolism	VQ scan	possible placement of
SpO ₂ 86%	Cough	Pericardial constriction	ABG	self-managed drainage
	Chest discomfort	Obstruction of	Thoracentesis provides	system for recurrent
	Abnormal breath sounds/	pulmonary vessels by	symptomatic relief	pleural effusions
	presence of pleural friction	tumor or stenosis	May be eligible for	Follow-up plan for early
	rub	Shedding of malignant	discharge if	diagnosis and
		cells into pleural space	thoracentesis provides	intervention for
			relief	recurrent pleural
			Cytological analysis of	effusion
			pleural fluid to	
			determine if malignant	
			cells are present	

Pneumonia [17, 18]. A 68-year-old man with prostate cancer presenting to ED with a productive cough, fever \times 3 days, shaking, and chills. He describes the sputum as thick and yellow. He also adds that a day ago, he developed pain in his right chest that is worsened with inspiration.

(continued)

Table 6.1 (continued)

Vital signs	Signs and symptoms	Risk factors/contributing factors	Potential tests/ interventions	Nursing considerations
BP 92/53 HR 120 T 39.0 R 26 SpO ₂ 90%	Fever, cough, sputum, hypoxemia, SOB Back pain, based on location of consolidation Vague, ill-defined symptoms Fatigue	Neutropenia Decreased activity Chronic obstructive pulmonary disease	General labs Blood cultures Lactic acid IV antimicrobial therapy Fluid resuscitation, ensure adequate cardiac status Sputum cultures Supplemental oxygen and use of noninvasive ventilation	Sepsis protocol Monitor changes in respiratory status, reposition as appropriate Maintain patent airway Promote normothermia Optimize fluid balance Encouraging coughing and deep breathing Promote adequate nutrition
Pneumothorax. A 68-year-old female with a large tumor in the right lung presents with sudden onset shortness of breath. She is in acute				

Pneumothorax. A 68-year-old female with a large tumor in the right lung presents with sudden onset shortness of breath. She is in acute distress and is breathing rapidly. Breath sounds are absent on the right.

BP 96/57	Signs/symptoms dependent on	Pulmonary malignancy	Chest X-ray/CT scan	Monitor changes in
HR 119	size/location of pneumothorax	Previous pneumothorax	Ultrasound	respiratory status
Т 38.6	Respiratory distress/failure	Procedures (i.e., central	Identification and	Assistance with chest
R 27	Dyspnea and chest pain	line insertion)	treatment of	tube insertion and
SpO ₂ 88%	Absent or decreased breath	Rupture of necrotic	underlying cause; may	management
	sounds on affected side	neoplastic tissue in	not require	Pain management
	Pneumothorax hyperresonant	pleural cavity [19]	intervention	
	by percussion	Tumor at lung periphery	Chest tube insertion	
	Deviation of trachea	[20–22]		
	Unequal chest expansion	Oncologic therapy [23]		
	Hypotension			

Deep vein thrombosis. A 54-year-old woman with uterine cancer and currently on treatment presents with left leg pain and swelling. She also reports that the swelling has been increasing over the course of 1 week. The affected leg is warm to touch, red, and edematous.

reports that the short hereasing over the course of 1 week the appendix to solver, rea, and each atoms				
BP 152/74	Tight ache, tight feeling, or	Procedures causing	General labs	Minimize or prevent
HR 74	frank pain in calf or behind	venous stasis (lengthy	Laboratory: D-dimer	respiratory compromise
Т 36.8	knee aggravated with standing	surgery)	Prothrombin/PTT	Understand condition
R 18	or walking; alleviated with	Active cancer (treatment	Doppler	(signs & symptoms),
SpO ₂ 96%	elevation	or palliation within	ultrasonography	risk factors, prevention
	Localized tenderness or pain	previous 6 months)	Treatment:	and management
	over involved vein	Hypercoagulable state	anticoagulants	Preventing further harm
	Tender, palpable venous cord	causing factors	Surgical intervention:	(adherence to treatment
	of involved vein	(physiological,	placement of vena	regimens, diet consistent
	Swollen calf or thigh by	environmental,	cava filter to prevent	with prescribed
	measurement	iatrogenic)	PE in recurrent DVT	medication, health
	Calf swelling more than 3 cm	Presence of intravenous	Low-molecular-weight	promotion)
	in circumference in	device (central venous	heparin (LMWH) may	Safety measures: avoid
	symptomatic leg	access device)	be self-administered	contact sports, use of
	Unilateral pitting edema in	Coagulopathy induced		soft toothbrush for oral
	involved extremity	by malignant cells		care and electric razor if
	Dilated superficial venous	Venous stasis (clot		there is a need to shave
	collateral vessels	formation; pooling)		Monitor for changes and
	(non-varicose)	Damage to blood vessel		report leg pain, bleeding
	Low-grade fever is possible	wall (endothelial)		or signs of
				thrombophlebitis or PE

Pericardial effusion and cardiac tamponade [24, 25]. A 55-year-old female with AML, currently on treatment, presents to the ED with complaints of a syncope episode with no injury. She endorses loss of consciousness for approximately "5 seconds" (verified by his adult son who was present at the time of fall). She reports that for the last 2 days, she has experienced increased fatigue associated with SOB. On assessment, her heart sounds are muffled on auscultation.

Table 6.1 (continued)

		Risk factors/contributing	Potential tests/	
Vital signs	Signs and symptoms	factors	interventions	Nursing considerations
Vital signs BP 108/60 HR 118 T 37.1 R 18 SpO ₂ 94%	Signs and symptoms Hoarseness, cough, hiccups, difficulty swallowing (compression of trachea, esophagus, vagal nerve) Muffled heart sounds Pericardial friction rub may be heard Increased jugular venous distension Kussmaul respirations Narrowing of pulse pressure – systolic blood pressure decreases and diastolic increases Paradoxical pulse (decline in systolic blood pressure on inspiration) Other signs of decreased cardiac output include tachycardia, anxiety, restless, peripheral cyanosis, oliguria, shock	factors Tumors most often associated with pericardial metastasis Some primary tumors (rare): sarcomas and mesotheliomas Lung and breast cancers can spread by direct extension of lymphatic metastasis Lymphomas and Leukemia routinely spread by hematogenous routes Radiation therapy of >4000 rad to the mediastinum Accumulation of excessive fluid within the pericardial sac (pericardial effusion) increasing pressure and	interventions Primary goal to remove fluid and relieve/prevent impending cardiac collapse Chest X-ray, CT ECG Echocardiography Percutaneous pericardiocentesis Pharmacologic management to control heart rate	Nursing considerations Patient teaching: early identification of signs/ symptoms Maximize safety with activities of daily life and ambulation Intervention to minimize severity: elevate head of bed; oxygenate; and manage pain and dyspnea Measures to enhance adaptation and rehabilitation
		compressing the neur		

Patients undergoing cancer treatment may also reduce their activity, placing them at higher risk of developing thrombosis, leading to pulmonary emboli. The patient may be asymptomatic but should be treated with daily injections of anticoagulants. It is important to evaluate the patient's clotting times and platelet counts, as this may exclude them as candidates for anticoagulation therapies [30].

Chief Complaint: Altered Mental Status

Altered mental status (AMS) is a frequent chief complaint in oncologic patients presenting to the ED [31]. These mentation changes can result from metabolic disturbances, structural changes (such as metastatic disease or intracranial hemorrhage), or infection [32]. Ruling out the most lifethreatening conditions is critical, as interventions are timesensitive and require prompt identification to achieve desirable outcomes [33]. Table 6.2 illustrates presentations involving altered mental status [34–45].

Patients may present with varying degrees of AMS based on causative factors [32]. They may present as confused, somnolent, inattentive, or with seizure activity, both focal and widespread. Consider the type of cancer, risk factors associated with metabolic changes, infection risk, metastatic disease, and bleeding risk. Associated presenting symptoms and vital signs will also assist in identifying the underlying cause. Obtaining a thorough history from a family member or caregiver may also provide relevant information to AMS's cause, including the onset of mentation change, medications, medical history, and significant events. While the presence of malignancy creates an increased likelihood of atypical differential diagnoses, it is important to consider still acute ischemic stroke, hypoglycemia, and other common underlying conditions for patients presenting with changes in mentation.

Most oncologic patients presenting to the ED with AMS should receive a STAT head CT to determine if there is hemorrhage, as oncologic patients on active treatment are at higher risk for thrombocytopenia leading to bleeds [33]. Additionally, patients with known brain metastases are at risk due to the highly vascular nature of neoplasia. In addition to diagnostic imaging, a CBC and comprehensive metabolic panel (CMP) should be obtained. Platelets and white blood cell count may indicate additional causes, such as bleeding or infection. Many treatments cause pancytopenia and electrolyte disturbances that may be relevant to the patient's condition. Disturbances in electrolytes, bilirubin, and ammonia may cause changes in mentation. For example, hypo and hypernatremia can cause significant mental status changes and a common metabolic disturbance in certain lung cancer types. Ammonia can also cause AMS and may be present in cancers with hepatic involvement. All these components of the initial workup will assist in identifying the cause of AMS.

In the presence of new metastases identified in diagnostic imaging, corticosteroids can reduce edema around the lesion and subsequently diminish AMS symptoms [46]. If an acute ischemic stroke is suspected, it is critical to verify the platelet count to determine if the patient is an appropriate candidate for

Table 6.2 Case studies: altered mental status

Vital signs	Signs and symptoms	Risk factors/ contributing factors	Potential tests/ interventions	Nursing considerations	
AMS related to metastatic disease [34]. A 50-year-old female with history of breast cancer with bone/liver metastases presenting with altered mental status. Her boyfriend states she has increased confusion over last few days. He states she is not answering questions appropriately and has not been taking her medications as directed. Patient oriented to person and place but does not know what year it is. She is inattentive and					
BP 117/68 HR 84 T 37.0 R 16 SpO ₂ 98%	Confusion (may be intermittent) Somnolence Seizures (may be focal)	Metastatic breast cancer Breast cancer has high metastatic risk in late stages	CT head Steroids (dexamethasone) Emergent neurosurgery to alleviate intracranial pressure if causing edema or ventricular obstruction	Frequent neuro vital signs Around the clock steroids to avoid additional edema Notify provider of any changes in mental status from initial baseline Elevate HOB and promote proper body alignment	
AMS related to metabolic disturbances – sodium brain. He presents with altered mental status exh but not time, GCS eye 3, verbal 4, and motor 6.	n [35–40]. A 56- nibiting delayed n	year-old male with small- responses, confusion, and	cell cancer of right lung, met decreased concentration. Or	astatic disease to bone/ iented to place/person	
BP 117/68 HR 124 T 37.0 R 16 SpO ₂ 98%	Confusion Decreased PO intake Fatigue	Small cell lung cancer Most common electrolyte abnormality with small-cell lung cancer is hyponatremia Advanced disease increases likelihood of metabolic complications	General labs Head CT scan r/o bleed or metastatic progression Sodium replacement	Rebound cerebral edema with hypertonic solutions Seizure precaution Frequent neuro vital signs to identify subtle changes	
AMS related to metabolic disturbances – ammonia [41–45]. A 49-year-old female with metastatic colon cancer presents with altered mental status. Oriented to self only. GCS eye 4, GCS verbal 5, GCS motor 6. Patient has a history of hyponatremia, hyperkalemia, liver metastases. Finger stick glucose 105 mg/dL.					
BP 93/58 HR 82 T 36.4 R 18 SpO ₂ 98%	Aggressive Confusion Lethargic Dehydration Hypotensive	Duodenum adenocarcinoma Cirrhosis of the liver Three cycles of chemotherapy Kidney damage and/or liver damage Drug, alcohol abuse Chemotherapy Colon cancer Liver failure	CBC, CMP, UA/C, PT/ PTT, ammonia, liver enzymes CXR, EKG, CT head to r/o bleed Fluid resuscitation to flush Lactulose	Place on seizure precautions Monitor cognitive facilities May need restraints if aggressive If patient is unconscious, may have to administer lactulose through NG tube or rectal Lactulose will induce diarrhea and can contribute to falls	

tissue plasminogen activator (tPA) [47]. If an infection is suspected, the patient should promptly receive broad-spectrum antibiotics. Timely administration of antibiotics can significantly improve patient prognosis in the presence of sepsis, with AMS being a frequent symptom indicating infection [48]. Patients with metabolic imbalances will improve upon the correction of the underlying disease process. Neurosurgery or neurology services may be consulted to address any neurological interventions based on ED findings [49]. Throughout the ED encounter, the EN should perform frequent neuro assessments to detect early deterioration signs. Placing the patient in semi-fowlers, elevating the head of the bed to 30 degrees or higher, and ensuring proper body alignment may also benefit patients with increased intracranial pressure [50]. Any changes in status should be immediately communicated with the provider. If steroids are ordered, they should be administered at scheduled times to reduce associated edema [49].
Chief Complaint: Back Pain

While back pain is a common chief complaint in EDs, the presence of back pain with cancer diagnoses can indicate metastatic spinal cord compression (MSCC), a time-sensitive emergency that requires prompt intervention. Although back pain is the most common complaint with MSCC, patients may also present with numbness, pain, or tingling in their extremities, bowel or bladder retention or incontinence, and even paralysis or gait disturbances. Patients with breast, lung, prostate, and renal cancer, as well as lymphomas and myelomas are at the highest risk, with men outnumbering women 2:1 [51]. See case study in Table 6.3 [51, 52].

Presenting symptoms will depend on the level of involvement and the degree to which the metastatic lesion is invading the spinal column. The degree of vertebral lesion invasiveness directly correlates with symptom severity. Symptoms may be alleviated with steroids by reducing the pressure on the spinal column [53]. Although symptoms may improve, these patients are at high risk for falls due to sudden sensory and motor function disturbances [54]. The patient's position may influence symptoms, activity level, level of involvement, and lesion location. Identifying a patient's position that reduces pain is important, and those with severe pain should be log rolled to avoid further injury. Range-ofmotion assessments should also be conducted with caution, as they can cause additional injury in the presence of osteolytic lesions. These patients are at high risk for spinal instability, pathological fractures, and caudal equine syndrome.

Assessing for urinary retention and post-void residual are also necessary to determine if urinary catheter place-

Table 6.3()	ase study:	back pair
-------------	------------	-----------

ment is necessary. Although patients may feel that they have fully emptied their bladder, there may be significant post-void residual, causing additional complications if not completely emptied. Patients may not state any bladder or bowel malfunctions due to loss of sensory perceptions, so assessment is necessary, regardless of patients' perceptions.

The radiological imaging modality of choice is magnetic resonance imaging (MRI) without contrast. If the patient cannot tolerate an MRI, a CT or X-ray may reveal findings but are not sensitive [55]. Treatments may include corticosteroids, radiation therapy, surgical intervention, and palliative chemotherapy [56]. Spinal cord metastases indicate late stages of cancer, and depending on patient functional status, treatment focus may be symptom management. Patients with MSCC may be candidates for advance care planning conversations, as this condition indicates advanced disease and poor prognosis [57].

Chief Complaint: Abdominal Symptoms

For patients with a concurrent cancer diagnosis, abdominal symptoms may indicate a variety of medical emergencies. Table 6.4 illustrates a number of presentations associated with these symptoms [58–80]. Oncologic patients are at high risk for bowel obstruction due to medication and antineoplastic treatments. Without prompt gastric decompression or surgical intervention, this can progress to perforation and severe infection. These patients may also present with nausea and vomiting due to the obstruction [81].

		Risk factors/	Potential tests/				
Vital signs	Signs and symptoms	contributing factors	interventions	Nursing considerations			
Metastatic spinal cord compres	Metastatic spinal cord compression [52]. A 67-year-old male with a history or metastatic prostate cancer presents to the emergency						
department with lower back pair	n 8/10 that has been prog	ressing over the last wee	k. He states he has some tir	ngling in his legs and feels weak			
when he is ambulating.							
BP 130/82	Primary	Men out-number	Diagnostics: radiological	Post-void bladder scan to			
HR 77	complaint = back pain	women 2:1 [51]	imaging of choice = MRI	evaluate for urinary retention			
Т 36.8	Weakness	Most prevalent in	without contrast	Avoid range-of-motion testing if			
R 16	Paraplegia	breast, lung, prostate,	CT or X-ray if patient	concern for spinal instability			
SpO ₂ 97%	Sensory disturbances	renal, lymphoma &	unable to tolerate MRI,	Best to immobilize patient as			
	(numbness,	myeloma	not as sensitive	much as possible to prevent			
	neuropathy)	Caused by vertebral	Assessments: serial	pathological fractures or			
	Autonomic	body metastasis	neurological evaluations,	additional pressure on the spinal			
	disturbances	invading the spinal	post-void bladder scan	cord			
	(incontinence, urinary	column	Treatments:	High fall risk			
	retention)	Level of involvement	corticosteroids, radiation	Strict bed rest for patients with			
		directly reflects	therapy, surgical	poor performance status or spinal			
		functional status and	intervention,	cord instability			
		clinical presentation	chemotherapy	Indicates advanced			
		Symptoms may be		disease = warrants advance care			
		affected by positioning		planning conversation			
		(i.e., sitting vs.					
		standing vs. laying					
		down)					

Table 6.4 Case study: abdominal symptom

5 5	1			
	Signs and	Risk factors/contributing	Potential tests/	
Vital signs	symptoms	factors	interventions	Nursing considerations
Bowel obstruction [58–61]. A 66-ye progressed and began 4 days ago. S. cancer. She has known metastatic pe for pain related to cancer pain and the	ar-old female with a he has a history of al eritoneal disease. Pat radiation.	history of ovarian cancer preser bdominal surgery, followed by ra tient has not had a bowel movem	nts with severe abdominal idiation and chemotherapy ent in 5 days and feels nat	pain (9/10). Pain has v treatment for her ovarian useated. Chronic opioid use
BP 146/90	Abdominal pain,	Previous abdominal surgery	General and	Monitor for changes, if
HR 106	cramping,	and scar tissue	coagulation labs	bowels perforate may
Т 37.8	distention	History of colon or rectal	Abdominal imaging	quickly progress to sepsis
R 22	Nausea and	cancer or from other organs	Nasogastric tube	Neuro checks
SpO ₂ 96%	vomiting	that has spread to the	insertion for	Support B/P
	Loss of appetite	abdomen	decompression	Possible sepsis
	Constipation and	Inflammatory bowel disease	Antibiotics, fluids, pain	NPO
	inability to pass	Diverticulitis	control	Pain management
	gas	Previous abdominal or pelvic radiation	Possible surgery	Strict I&O (Foley) Fall risk
		Radiation and previous		
		abdominal surgery		
		Opioid induced constipation		
		Age		
		Intra-abdominal lesions and		
		surgical scarring		

Diverticulitis [62, 63]. A 52-year-old white female history of melanoma, diverticulosis, and constipation with recent chemotherapy. The patient complains of abdominal pain in the lower left side over the past week progressively getting worse last night. Her pain level is 7/10.

1 5 1	0	1 1 0 20	0 0 1	
BP 160/88	Pain lower left	Advanced age	CT scan of the	Pain management
HR 94	side of abdomen	Obesity	abdomen and pelvis,	Hydration
Т 38.0	progressively	Smoking	CBC, chemistries	GI rest clear liquids
R 20	getting worst over	Diet high in animal fat and	Rest, oral antibiotics,	Nutrition education
SpO ₂ 97%	the last 5 days	low in fiber	liquid diet	high-fiber diet, starting
	Nausea and	Certain medication	More severe IV	with low fiber initially
	vomiting	Genetics	antibiotics, hospital	
	Fever	Diverticulosis	admission, surgery	
	Abdominal	Immunocompromised	Mild case may be	
	tenderness,	Constipation	discharged home if able	
	cramping		to tolerate PO	
	Constipation			

Gastrointestinal bleeding [62, 64]. A 72-year-old Hispanic male with esophageal cancer presents to the ED with abdominal pain over the last month and hematemesis this morning, reports black tarry stools and weakness progressing over the last week. History of pulmonary embolism 2 months ago, on coumadin.

Hematemesis	History of peptic ulcer disease	CT scan of abdomen	Assess for bleeding in
Black tarry stool,	or GI bleed	and pelvis with IV	stool
rectal bleeding in	Advanced age	contrast	ECG
or with stool	NSAID, anticoagulants	CBC, PT with INR,	Strict I&O
Abdominal pain	Esophagitis	PTT, D Dimer,	Administer pantoprazole
Weakness	IBD, colon polyps,	fibrinogen, type and	Monitor heart rate and
Low blood	hemorrhoids, diverticular	screen, CMP,	blood pressure
pressure	disease, proctitis, anal fissures	magnesium,	Monitor H&H and clotting
	Esophageal tumor	phosphorus, amylase,	times
		lipase, UA, urine	Assess patient history and
		culture	medications
		EKG 12 lead, FOBT,	Support
		endoscopy,	
		colonoscopy,	
		angiography	
		Cardiac monitoring	
		May be given PPI, may	
		be taken off blood-	
		thinning medications,	
		pain medication	
	Hematemesis Black tarry stool, rectal bleeding in or with stool Abdominal pain Weakness Low blood pressure	HematemesisHistory of peptic ulcer diseaseBlack tarry stool, rectal bleeding in or with stoolor GI bleedAbdominal painAdvanced ageWeaknessIBD, colon polyps,Low bloodhemorrhoids, diverticularpressuredisease, proctitis, anal fissuresEsophageal tumor	HematemesisHistory of peptic ulcer diseaseCT scan of abdomenBlack tarry stool, rectal bleeding in or with stoolor GI bleedand pelvis with IV contrastAbdominal painAdvanced agecontrastWeaknessIBD, colon polyps, hemorrhoids, diverticularfibrinogen, type and screen, CMP,pressuredisease, proctitis, anal fissures Esophageal tumormagnesium, phosphorus, amylase, lipase, UA, urine cultureEKG 12 lead, FOBT, endoscopy, colonoscopy, angiographyCardiac monitoring May be given PPI, may be taken off blood- thinning medications,

Table 6.4 (continued)

V 7/4-1-1-1-00	Signs and	Risk factors/contributing	Potential tests/	Numine and the discussion
vital signs	symptoms	Tactors	interventions	Nursing considerations
Diarrhea [65, 66]. A 35-year-old As	ian female with brea	st cancer presents with 2 days of	f watery stool, abdominal of	cramping, and low-grade
fever. She received chemotherapy ap	proximately 3 days a	ago and feels weak and exhauste	d. Her primary oncologist	recommended she come to
the ED.				
BP 102/58	Abdominal pain	Viruses, bacteria, parasites	CBC, CMP,	Assess for abdominal
HR 107	Watery stool	Medications, including	magnesium,	discomfort, loose stools,
Т 37.9	Fever	chemotherapy	phosphorus, stool	cramping
R 18	Abdominal	Graft versus host disease	culture, stool for C-diff,	Inquire about: tolerance to
SpO ₂ 96%	distention	Lactose intolerance	FOBT, colonoscopy	milk and other dairy
		Surgery	Antibiotics, adjusting	products, food preparation,
		Recent chemotherapy	medications being taken	medications patient is or
		Infection	Treatment plan to	has been taking, and
		Food contamination	replace lost fluids and	current stressors
			electrolytes	Check for history of
			Observation pending	abdominal radiation, GI
			test results	diseases, foreign travel,
			May discharge home	and drinking untreated
			after hydration and	water
			diarrhea resolves	

Constipation [67, 68]. A 76-year-old male with sigmoid adenocarcinoma presents seeking treatment for increasing abdominal pain and constipation persisting for 3 weeks. He is on Oxycontin. The patient complains of passing dry, hard stool every 5 days and a desire to defecate. Strains without relief after having a bowel movement.

<i>, , , , , , , , , ,</i>				
BP 168/88	Dry hard stools	Age	CBC, chemistries	Classify medication usage
HR 86	Passing fewer than	Diet low in fiber	CT of abdomen and	that may lead to
Т 36.9	3 stools a week	Little to no physical activity	pelvis, colonoscopy,	constipation
R 18	Straining to have	Taking certain medications	X-ray, anorectal	Assess patient's activity
SpO ₂ 98%	bowel movements	including sedatives, opioid	manometry,	level
	Abdominal pain	pain medications,	defecography	Assess patient's diet and
		antidepressants or medications	(outpatient)	eating habits
		to lower blood pressure	Increase fiber intake,	Check frequency and
		Cancer	increase exercise,	consistency of stool
		Poor hydration	prescription medication	Check for history of
			and laxatives	neurogenic diseases
			Surgery	
			Admit to observation	
			Discharge home if able	
			to provide relief with	
			enema/medication and	
			CT negative	

Nausea and vomiting [69–71]. A 40-year-old African American female with uterine carcinoma on active treatment presents to the ED complaining of nausea and vomiting. She has vomited 4 times in the last hour and is unable to keep anything down orally. Her nausea is increased with certain smells.

BP 95/60 Na HR 116 voi T 37.1 We R 20 fat SpO ₂ 96%	ausea and omiting Veakness and tigue	Cancer treatment Emotional distress Medication Gender BMI Motion sickness History of migraine Tumor Obstruction	CT abdomen/pelvis w/ contrast General/abdominal labs EKG, cardiac monitoring Clear liquids IV hydration Electrolyte replacement Antiemetics Patient may be managed at home with instructions and antiemetics as needed if improves and test are negative	NPO, may progress to clear liquids with PO challenge Assess medications that may lead to nausea/ vomiting Asses abdomen for distention and cramping, frequency of vomiting and emesis contents Strict I&O Fall risk
--	---	---	--	--

(continued)

Table 6.4 (continued)

Vital signs	Signs and	Risk factors/contributing Potential tests/		Nursing considerations
Urinary retention [72–76]. A 73-yea	ar-old male with blac	dder cancer and suprapubic cath	eter presents with urinary	retention for 3 days and
abdominal pain worsened by tactile	pressure.	I I	1	
BP 166/90 HR 106 T 36.8 R 20 SpO ₂ 94%	Acute suprapubic pain Anuria Distended bladder Urgency	Benign prostatic hyperplasia Bladder cancer Hemorrhagic cystitis History of hypertension and diabetes mellitus Increased age Affects men more than women Postop complication (s/p TURP) Medication related Blockage (stone, mass) Urinary tract infection Abscess Inflammation (cystitis, urethritis) Pelvic radiation Cord compression Penile trauma Fecal impaction	Bladder scan, CBC, CMP, UA/UC, indwelling catheter placement Medication for bladder spasm (hyoscyamine) Antibiotic for UTI Urology consult	Assess for previous surgeries/trauma/tumor Place catheter, preferably 16Fr or large enough to pass blood clots for that is determined to be the issue, may have to use coudé tip if patient has enlarged prostrate Consult urology if unsuccessful with catheter placement Monitor patient for electrolyte abnormalities, dehydration, hypotension after rapid bladder decompression Maintain adequate fluid intake
Acute kidney injury [77]. A 65-year abnormal elevation of creatinine fro	old male with prost of routine office appe	ate cancer and chemotherapy 2 c pintment.	lays ago presents to the E	D with asymptomatic
BP 122/80 HR 76 T 36.9 R 16 SpO ₂ 97%	Leg swelling Potassium 6.7 Creatinine 8.32	Bladder cancer Nephrotoxic medications, including chemotherapy Obstructive hydronephrosis (tumor, clot)	Repeat labs, CBC/ CMP, BUN UA, serum and urine electrolytes EKG and cardiac monitoring Renal ultrasound Fluid resuscitation Renal consult Possible surgical intervention (percutaneous nephrostomy) Kayexalate, albuterol nebulizer, 10 units insulin, calcium gluconate, 1 amp, D25, bicarb 50 meq	Review medications to discontinue nephrotoxic medications Anticipate adjusted medications according to renal function Monitor pulmonary and cardiovascular events due to fluid overload and electrolyte imbalances Monitor I/O Monitor changes in mental status Complications of acute kidney injury in cancer patient may limit the patient's ability to continue treatment Monitor blood glucose before and after insulin dose
Hematuria. A 79-year-old male with in urine for the past 12 hours. The p BP 164/74 HR 90 T 36.5 R 20 SpO ₂ 90% (on 2 LPM via nasal cannula)	a bladder cancer and atient was seen at ar Hematuria Blood clots Urinary retention Pain	t renal cancer presents to the ED a outside facility and found to ha Bladder, urethral or kidney cancer UTI Trauma (pelvic area, renal) Hemorrhagic cystitis Pelvic radiation Chemotherapeutic agents (ifosphamide, cyclophosphamide) Medications Nephritis Calculi Renal cysts Enlarged prostate (causing to strain and rupture vessels)	with complaint of lower of ve creatinine 2.4, pyelone ₁ Labs (CBC, CMP, UA/ UC), adequate fluid intake Continuous bladder irrigation Diagnostic imaging (renal U/S, cystoscopy) Antibiotics for treatment of UTI Urology consult	abdominal pain and blood phritis, and cystitis. Bladder irrigation via 3-way catheter, titrate drip to light pink, almost clear output, continuous irrigation If interrupted, clot may form If leaking at catheter insertion site, catheter most likely blocked with blood clot and clot will need to be irrigated May have to use coudé tip catheter if patient has enlarged prostate. Monitor hemoglobin and electrolytes

Table 6.4 (continued)

Vital signs	Signs and symptoms	Risk factors/contributing factors	Potential tests/ interventions	Nursing considerations	
Bile duct obstruction [78–80]. A 72-year-old female with pancreatic cancer on active treatment presents to the ED with vomiting and fatig for the past week. She is jaundiced and slightly confused. She states she has generalized pruritus and abdominal cramping.					
BP 104/66 HR 76 T 36.9 R 16 SpO ₂ 99%	Projectile vomiting Upper right abdominal pain Lethargy, anorexia/decrease in appetite Severe heartburn/ reflux	Pancreatic cancer Female Increased age Diabetes mellitus Type II	General labs, bilirubin, alk phos, liver enzymes CT scan GI endoscopy consult for stent placement or G-tube placement	If undergoing biliary drainage patient should receive broad-spectrum antibiotics within 1 hour of start of procedure due to transit bacteremia during or after the procedure Monitor for bleeding, leakage around the tube and subsequent skin breakdown, catheter related pain, pancreatitis, sepsis	

Cancer-related treatments may also cause acute kidney injuries (AKI) present as abdominal pain, oliguria, and flank pain. Depending on cancer location and gastrointestinal involvement, disease progression may be the primary factor causing pain or obstruction. Location of the pain, severity, onset, aggravating and alleviating factors, as well as medical and oncologic history is important in determining the cause of abdominal pain and necessary interventions. Due to pancytopenia caused by many treatments, bleeding and infection should also be considered if indicated in clinical presentation [81].

Unfortunately, cancer treatment frequently causes nausea and vomiting. Prevention of dehydration and symptom management are most important in chemotherapy-induced nausea and vomiting and depending on the severity, patients may require scheduled administration of multiple antiemetics. Sensations, including smell and taste, are also impacted with chemotherapies, and something as innocent as perfume may trigger emesis. Nurses should avoid wearing any creams, lotions or perfumes with strong scents that may trigger episodes of nausea and vomiting [82]. When attempting oral intake, small volumes of plain food and drink are best, as foods with strong flavors or smells may also increase the risk for emesis or even aspiration. Additionally, elevating the head of the bed to prevent aspiration is an important safety measure, as vomiting episodes may be sudden without warning [83].

In addition to upper GI symptoms, cancer treatments can cause lower GI symptoms such as constipation and diarrhea. Severe constipation can develop with both treatment and symptom management therapies, such as opioids for pain management. All cancer patients should be on a stool softener to prevent fecal impaction that can lead to additional complications. Before administering an enema, platelet levels should be verified to ensure no bleeding risk. In cases of diarrhea, dehydration can quickly progress and electrolyte imbalances may occur. Prompt replacement of fluids and electrolytes is necessary to prevent further complications related to electrolyte deficiency. While treatments can induce adverse events, patients with previous stem-cell transplants may experience similar symptoms due to graft-versus-host disease (GVHD). In these cases, tacrolimus levels should be monitored and steroids are generally the treatment of choice [84].

Ascites and abdominal distention are commonly seen in patients with metastatic peritoneal disease. Ascites may be recurrent and require frequent removal of peritoneal fluids via paracentesis. For patients with recurrent ascites due to metastatic disease, a peritoneal drain may be indicated to allow the patient to self-drain fluid build-up in the abdomen and prevent frequent ED visits. Patient education and discussion with the primary oncologist will help determine if the patient is an appropriate candidate for peritoneal catheter placement [85].

With cancer patients being at high risk for infection, the presence of colitis, gastritis, and diverticulitis should also be assessed to determine if a patient requires antibiotic therapy. Infection should always be addressed in any cancer patient presenting with abdominal symptoms to prevent further deterioration [81].

Abdominal symptoms are common for both general ED patients and cancer patients and are caused by various conditions. The patients' medical and oncological history can guide patient diagnoses, including cancer type and associated events leading up to the ED encounter. Evaluation of laboratory findings, including hepatic functions, pancreatic enzymes, CBC and CMP, is also essential in determining an appropriate treatment course.

Chief Complaint: Infection

Many cancer patients undergoing treatments experience pancytopenia, including neutropenia. This places them at significantly higher risk for developing an infection and becoming septic. Patients may present initially with a fever and neutropenia and otherwise stable vital signs. That said, these patients have a minimal metabolic reserve and no immune defense mechanisms, so they can quickly decline without the initiation of appropriate interventions. A central line is frequently standard in patients receiving chemotherapy regimens. This direct access to the bloodstream also places patients at higher risk for bacteremia and sepsis [86].

Development of a sepsis protocol and standing parameters for early interventions can help decrease the time from door to antibiotic administration, resulting in more favorable outcomes. Once the infection source is identified, antibiotic therapy should be tailored based on the organism's susceptibility to promote antibiotic stewardship [87].

Patients presenting with fever can be quickly identified as having a potential infection or sepsis. Some patients experience a condition called "tumor fever," which is the most frequent cause of pyrexia unrelated to infection. This is most commonly present in leukemias, lymphomas, sarcomas, renal cell carcinomas, and patients with liver metastases, but may present in any type of cancer. Although the cause may be unknown, patients should be treated as if an infection is present until otherwise ruled out [88].

There are some cases where patients are afebrile but exhibit tachycardia, tachypnea, or hypotension. These may indicate infection but are nonspecific in cancer patients. These could be caused by many other conditions common in cancer patients, including anemia, dehydration, or different physiological responses to malignancy. Infection is frequently the culprit in these cases, but does not rule out other diagnoses. Additionally, ED nurses should identify if the patient is taking any medications that could reduce the temperature before ED arrival, such as acetaminophen or ibuprofen. This may mask the fever and cause infection to be overlooked [88]. Table 6.5 illustrates typical presentations for neutropenic fever and sepsis [89–93].

Chief Complaint: Newly Diagnosed Cancer in the Emergency Department

Although less common, patients may present to EDs without a cancer diagnosis, only to be diagnosed during treatment in the ED [94]. These situations can be high acuity and high stress, as the patient's medical management may be complicated, and the emotional stress of the patient and family will likely be heightened. The initial presentation will vary based on underlying cancer diagnosis but may range from nonspecific complaints to a growing tumor site. Regardless of the final cancer diagnosis, the patient and family will need significant psychosocial support to begin their journey as cancer patients [95].

For patients without established primary care, the ED may be their only access to medical services. Unfortunately, patients presenting with an invasive solid tumor without

Table 6.5 Case studies: infection

Signs and
symptomsRisk factors/
contributing factorsPotential tests/
interventionsNursing considerationsNeutropenic fever [89]. A 52-year-old female currently undergoing
origin. History of stage III breast cancer; treatment or pleted 2 years ago. White blood cultures
BP 98/68Potential tests/
interventionsNursing considerationsBP 98/68Fever
FatiguePoor performance
statusCBC with diff
antibiotics to prayent further

Fever	Poor performance	CBC with diff	Timely administration of
Fatigue	status	Blood culture	antibiotics to prevent further
	Advanced oncologic	Broad-spectrum	deterioration and sepsis cascade
	disease	antibiotics	
	Low blood cell		
	counts		
	Chemotherapy		
	Hematologic cancers		
	Opportunistic		
	infection		
	Fatigue	Fever Poor performance Fatigue status Advanced oncologic disease Low blood cell counts Chemotherapy Hematologic cancers Opportunistic infection	FeverPoor performanceCBC with diffFatiguestatusBlood cultureAdvanced oncologicBroad-spectrumdiseaseantibioticsLow blood cellcountsChemotherapyHematologic cancersOpportunisticinfection

Sepsis [90–93]. An 86-year-old female presents with fever, SOB, fatigue. She states she has back pain worsening over past 24 hours. Had chemotherapy for Stage III breast cancer approximately a week and a half ago. She has a double lumen peripherally inserted central catheter (PICC) line in her right arm.

(1100) the http://www.				
BP 77/47	Hypotension	Central venous	CBC with diff	Sepsis mortality increases
HR 127	Tachycardia	access	Blood cultures	Poor tissue perfusion
Т 38.9	Fever	Neutropenia - nadir	Lactic acid	
R 24		from chemotherapy	Broad-spectrum	
SpO ₂ 89%		Increased age	antibiotics	
		Chemotherapy		
		Hematologic cancers		

prior cancer diagnosis tend to have a poor prognosis. Patients diagnosed with cancer in the ED are usually in late stages of the disease, with a 75% higher risk of being diagnosed as stage 4 cancer, versus stage 1 [96]. These patients may not have access to healthcare in a primary care setting, causing them to utilize the ED for access to treatment. Patients may not have symptoms severe enough to prompt an ED encounter until cancer has progressed. Delays in cancer diagnosis significantly increase the likelihood of metastatic disease being present upon initial diagnosis. Because of the poor prognosis, advance care planning should be discussed with the patient and family to ensure the quality care that supports the patient and family's wishes. Generally, solid tumor patients will not require immediate antineoplastic therapy in an emergency setting. ED nurses should focus on symptom management, oncologic plan of care post-ED visit, and appropriate supportive services such as social work, case managers, nutritionists, and pain management specialists [96].

In contrast with solid tumors, different types of acute leukemias may present as a medical emergency and require prompt cancer treatment and immediate antineoplastic therapies. These cancers are frequently diagnosed in the ED and are sometimes only identified via blood work. Patients with acute leukemias tend to have nonspecific symptoms of infections. The proliferation of immature white blood cells causes insufficient immune defense mechanisms required to fight an infection, and this may be the only indication of the underlying issue. An analysis of blood work and subsequent bone marrow biopsy will identify the specific cancer type and appropriate treatments. These patients are at high risk for sepsis due to an ineffective immune system and coagulopathies due to increased blood viscosity and thrombocytopenia secondary to leukemia [97].

Patients with newly diagnosed leukemia should be regularly monitored for status changes, as they can quickly deteriorate. Those with a white blood cell count of 50,000 or greater will require immediate therapy to reduce the number of immature blasts in circulation. The type of cancer will determine the induction phase of treatment, which the ED nurse will likely initiate. Due to the high-risk and timesensitive nature of induction therapies, protocols should be developed by both emergency and oncologic departments to ensure there are no administrative barriers that may prevent the patients from receiving immediate induction therapy in the ED. This collaboration between these specialties can be a significant factor influencing the patient's care course and subsequent outcome [97].

The ED nurse's role in newly diagnosed cancer patients is essential for their quality-of-life trajectory. Ensuring adequate education and resources can completely alter the patient's experience in the presence of a life-changing diagnosis, such as cancer. Although not all patients will be candidates for curative therapies, providing patients with all potential treatment options and plans of care will ensure they are on track for course best suited to their medical and psychosocial needs. Along with coordinating the various services, including patients and families, the care team should be prioritized by the ED nurse once medically stabilized [97]. Table 6.6 illustrates presentations for newly diagnosed acute myeloid leukemia and acute promyelocytic leukemia [98–105].

Chief Complaint: Malignancy Progression, Antineoplastic Treatments, and General Medical Emergencies

Complications may arise with patients throughout the course of their cancer diagnosis. These may arise both from disease progression and impact on physiological processes and antineoplastic therapies and the associated adverse effects. These can range from mild to severe, and a thorough patient medical history can help determine the underlying cause. As previously mentioned, treatments can cause pancytopenia resulting in anemia, thrombocytopenia, and neutropenia. These can lead to more severe complications such as infection or bleeding if not promptly addressed with the appropriate replacement or supportive therapy [106].

Disease progression will generally be related to the tumor's location and associated symptoms. Patients with primary or metastatic osteolytic lesions may develop pathological fractures as the disease progresses. These osteolytic lesions may initially cause the patient mild-to-moderate pain at the tumor site with an acute event producing a pathological fracture [107]. Patients with large abdominal or thoracic tumors can develop superior vena cava syndrome as the tumor grows, placing pressure on blood's systemic return to the heart. Certain neuroendocrine tumors can cause significant disturbances in hormonal and metabolic function, possibly resulting in diabetes insipidus, acute adrenal crisis (Addisonian crisis) and hypophysitis [108]. The tumor location and associated symptoms will greatly assist with determining differential diagnosis and appropriate treatments.

While cancer itself can produce adverse events seen in an emergency setting, the treatments patients receive can also have severe therapy-related adverse events. Patients who have received a stem-cell transplant (SCT) may also present with graft-versus-host disease (GVHD) complications. These can affect all organs and systems and are frequently treated with high-dose steroids [109]. Along with pancytopenia and associated conditions, many chemotherapies are nephrotoxic and cardiotoxic. Depending on the patient's ability to tolerate the treatment and underlying comorbidities, some patients may have more severe reactions than others [110]. Reviewing the patient's wallet insert that identifies
 Table 6.6
 Case studies: newly diagnosed cancer in the ED

, <u>,</u>				
Vital signs	Signs and symptoms	Risk factors/	Potential tests/	Nursing considerations
vital signs	Signs and symptoms	contributing factors	Interventions	Nurshig considerations
Acute myeloid leukemia (AML) [98–102]: A instructed his to present to the ED. The patie prescribed PO antibiotics and obtained a CB Hgb 6.1. The patient is a smoker (1/2 per day BP 117/76 HR 107 T 37.7 R 18 SpO ₂ 97%	A 32-year-old male presents int had an upper respirator BC. The patient has no sign y) and works at a chemical Frequently present after unresolved infection and discover abnormal blood counts May have weight loss, bleeding/bruising and fatigue due to counts	s with abnormal lab res y infection for 2 weeks ificant medical history. plant. AML more common in males than females Risk factors for AML are exposure to certain chemicals (work exposure), radiation and smoking	sults from his primary that prompted him to CBC results showed General labs Hydroxyurea for leukocytosis Bone marrow biopsy PRBC and platelet transfusions to replace counts Chest X-ray to evaluate for potential pulmonary infiltrates Antibiotic and antiviral therapy to prevent infectious complications	<i>care provider who</i> <i>see her PCP. PCP</i> <i>WBCs 173 k, Platelets 4 k,</i> High-risk for bleeding with thrombocytopenia Disseminated intravascular coagulation Highly viscous blood due to increased WBCs At risk for tumor lysis syndrome due to systemic cancer involvement
increased bruising. The patient has worked a severe neutropenia, anemia, and thrombocyte	nt a crude-oil processing pl processing pl	ant for 22 years and he	as no significant heali	th history. Lab results show

evere neuropenia, anenia, ana interiore	ytopenta			
BP 110/68	Anemia	Middle aged	General labs	High-risk cancer, but
HR 96	Thrombocytopenia	Long-term exposure	All-trans-retinoic	highly curable if treated
Г 37.2	Neutropenia	to petroleum	acid (ATRA)	timely
R 20		products	Bone marrow	Extremely rare subtype of
SpO ₂ 99%		Unknown	biopsy	AML
		May be associated	PRBC and platelet	Requires pathology
		with work exposure	transfusions to	evaluation of cell
			replace counts	morphology to determine
			Chest X-ray to	if APL versus other types
			evaluate for	of acute leukemia
			potential	
			pulmonary	
			infiltrates	
			Antibiotic and	
			antiviral therapy to	
			prevent infectious	
			complications	

the antineoplastic agent they are receiving may be beneficial. Obtaining this information as early as triage can help determine the differential diagnosis as well as medical management [110].

Dedication of resources to cancer treatment research has led to new therapies, emerging as first-line treatments producing promising outcomes. One of these recent advances is the increase of immunotherapies. Although these generally have a lower risk of associated adverse events, immunotherapies such as chimeric antigen receptor (CAR) T-cell therapy can induce specific emergent conditions, most frequently being cytokine release syndrome (CRS) and CAR-related encephalopathy syndrome (CRES). These emergent conditions may develop after receiving immunotherapies and must be treated timely to prevent long-term deficits. CRS and CRES have specific grading systems that should guide medical management and determine the severity of the condition [111].

Patients with cancer may have comorbidities, such as diabetes, hypertension, psychiatric conditions, cardiac dysrhythmias, and other chronic illnesses, and patients may present with conditions completely unrelated to their cancer or treatment. General medical emergencies such as ischemic stroke, myocardial infarction, or diabetic ketoacidosis (DKA) should still be considered, even in the presence of concurrent cancer, if the clinical presentation is consistent with the noncancer-related condition [112]. Notwithstanding, precautions should be taken when determining therapy for the medical emergency and how the patient's cancer may impact the typical course of management. For example, patients exhibiting signs of ischemic stroke may not be candidates for tissue plasminogen activator (tPA) based on platelet count, coagulation studies, and bleeding risk. All factors should be considered when determining medical management for the patient. The ED nurses' role is imperative to ensure a holistic approach to patient care [113]. Table 6.7 illustrates the wide variety of patient presentations discussed above [114–144].

Chief Complaint: End of Life in Advanced-Stage Cancer

Although advances in oncologic treatments have greatly improved the overall survival rate of cancer, end of life, patients with advanced cancer disproportionately represent cancer-related visits to emergency departments. The highacuity and fast-paced environment in the ED has been conventionally felt to be incompatible with end-of-life (EOL) discussions. The delicate topic is rarely addressed in EDs and can increase the psychosocial burden on the patient and family, increased costs, and futile care initiation. By providing a holistic approach and initiating conversations to establish care goals, the ED can help enhance the value and quality of care for EOL cancer patients [13]. Table 6.8 describes a typical presentation for a cancer patient presenting to the ED at the end of life [145–147].

Although the ED is not typically a setting where EOL discussions occur, initiating a palliative care (PC) consultation to assist with determining care goals with the patient and family member can greatly assist ED personnel in navigating the complex sequelae of the dying process in all domains [13]. Patients and families will require substantial physical, emotional, and spiritual support to ensure a smooth transition to hospice care. The ED nurse plays a central role in coordinating this care by ensuring all necessary services can provide expertise and guidance for the multifaceted needs associated with dying. These will include palliative, pain, hospice, nutrition, social work, case management, chaplaincy, and other multidisciplinary services based on the patient and family's unique needs. This experience can be the difference between a traumatic and peaceful death for both the patient and their loved ones [13].

As a gateway to hospitalization, the ED plays a vital role in the quality and value of EOL cancer patients' care. It is the tendency of ED personnel to choose life-saving interventions over meaningful conversations about advance care planning (ACP). EDs should ensure proper training and education are provided to staff to provide quality care and ensure dignity, compassion, and comfort for EOL patients. Altered mental status, dyspnea on minor exertion, and poor performance sta97 Idies to be the

tus (ECOG 3 or 4) were found in previous studies to be the "Triple Threat" predictors of mortality in advanced cancer. Patients with two or more of these conditions had a predicted 30-day mortality of 49% (95% CI. 34%, 64%) [145]. This may be used as a triage screening tool to identify advanced cancer patients who may benefit from care goals conversation.

In 2014, the Institute of Medicine (IOM) released the report, "Dying in America," calling for significant reform of the healthcare system to improve the quality and value of EOL care in America. The report cited recommendations to improve EOL care, including patient-centered and family-oriented EOL discussions, professional education and development of palliative care, healthcare policies to support EOL initiatives, and public education and engagement [148].

The National Quality Forum endorses multiple ED visits within the last 30 days of life to indicate poor-quality cancer care. Additional indicators of poor-quality cancer care include admission to the intensive care unit (ICU) within the last 30 days of life, death in the ICU, curative chemotherapy treatment in the last 14 days of life, and hospice admission for less than 3 days before death. ED nurses should advocate for EOL cancer patients to avoid poor-quality outcomes, enhance care value, and provide a positive experience for patients and families [148].

Ideally, these conversations would be initiated in an outpatient setting allowing ample time for discussion between providers, patients, and families. However, this is not always the case, and patients may only find out they are dying upon presentation to the ED. Although it may be perceived as challenging or inopportune, initiating a discussion of care goals in the ED can be one of the greatest gifts a nurse can provide to a patient [149].

Pandemic Response: SARS CoV 2 – Novel Coronavirus

Novel coronavirus disease (COVID-19), also termed SARS-COV-2, has emerged as a global threat and healthcare concern [150]. The virus first cases were reported in Wuhan, China, and marked the beginning of a global pandemic that completely upended daily life and the world's healthcare system [151]. Human-to-human transmission of COVID-19 occurs via respiratory droplets (by coughing or sneezing) and through direct contact with infected individuals or indirect contact with fomites of the affected individuals' environment [150]. Since its outbreak in China at the end of 2019 and until the April 5, 2020, the pandemic has affected > a million persons and caused 62,773 deaths worldwide [152].

		Risk factors/	Potential tests/	Nursing
/ital signs	Signs and symptoms	contributing factors	interventions	considerations
Cytokine release syndrome (CRS) & CA	R-related encephalopathy s	yndrome (CRES) [114]. A	A 61-year-old male preser	nts to triage with
achycardia, hypotension, and shortness	of breath. He received a chir	neric antigen receptor T-c	ell (CAR-T-cell) therapy	infusion for treatment
f lymphoma approximately 3 days ago a	nd was instructed to present	t to the ED.		
3P 88/54	Looks a lot like sepsis	Recent CAR-T cell	Tocilizumab is first	CRS & CRES
IR 126	Fever, myalgias,	therapy infusion	line treatment	grading system
37.2	anorexia, evidence of	Liquid tumor, potential	Supportive therapy	Assess for CRES in
20	multiple organ	high tumor burden can	Rule out infection	the presence of CRS,
pO ₂ 89%	involvement (dyspnea,	cause increased		may be concurrent
-	hypotension,	cytokine release		Maintain $SpO_2 > 92\%$
	arrhythmias, confusion	A therapy-induced		
	seizures)	immune systemic		
		reaction		
		Release of IL-6		
		proteins causes		
		systemic inflammatory		
		process		

Table 6.7 Case studies: malignancy progression, antineoplastic treatments, and general medical emergencies

Pathological fractures [115–117]. A 55-year-old female with metastatic breast cancer with known bone metastasis. She has been experiencing right groin pain that started approximately 3 months ago and got significantly worse over the last 3 days. The patient is experiencing severe pain and is unable to bear any weight on her right leg, prompting her to present to the ED. The patient is slightly tachycardia (112), all other vitals are WDL.

BP 130/88 HR 112 T 37.2 R 18 SpO ₂ 95%	Pain, sometimes chronic with acute exacerbation Acute change in functional abilities of affected limb	Metastatic breast cancer to the bones Weight bearing activities on a bone that has a metastatic lesion	X-ray to evaluate for acute pathological fractures Orthopedic consult to evaluate for possible surgical reconstruction if patient is a candidate Pain control	Log-roll patients to avoid further injury Premedicate with analgesia prior to movement Stabilize with pillows to avoid positional exacerbation Be aware of other bone lesions and take extra precautions as appropriate
---	---	--	---	--

Diabetic Ketoacidosis (DKA) [118–120]. A 64-year-old male undergoing treatment for prostate cancer with Lupron and prednisone. He presented to the ED with chief complaints of progressive weakness, confusion, loss of appetite, and nausea. The patient was diagnosed with new onset diabetes presenting with DKA, hyperglycemia, and acute kidney injury. Glucose 520 mg/dL on serum chemistry, bicarb 17, and anion gap 2

anion gap 2.				
BP 114/74	Glucose greater than	Diabetes type 1	Urinalysis and culture	Fluid volume status
HR 102	250 mg/dL	Diabetes type 2	Serum ketones	Increased risk for
Т 37.3	Dry mouth, dry skin	Long-term steroid use	Arterial blood gas	infection
R 20	Polyuria, polydipsia,	Non-compliance with	Blood culture	Knowledge deficit
SpO ₂ 94%	polyphagia	insulin therapy	CBC and chemistry	regarding glucose
	Changes in mentation	Infection	Cardiac monitoring	management
	Kussmaul respirations	Trauma	Neuro assessments	
			Critical care	
			Insulin therapy,	
			hydration, electrolyte	
			replacement	
Addisonian crisis [121, 122]. A 67-year-ol	ld undergoing treatment for	recurrent metastatic uter	ine leiomyosarcoma. She	presented with fatigue
and altered mental status. The patient was	diagnosed with adrenal ins	ufficiency secondary to hy	pophysitis following imm	nunotherapy.
BP 78/54	Tachycardia	Addison's disease	Serum cortisol level	Fluid and electrolyte
HR 116	Altered mental status	Prolonged	Chemistries	management
Т 36.2	Dry skin	administration of	High-dose IV	Fall precautions
R 22	Hypotension	glucocorticoids	corticosteroid therapy	
SpO ₂ 96%	Low fasting blood	Infection	Cardiac monitoring	
	glucose	Cancer	ACTH stimulation test	
		Stress	(cosyntropin)	
		ACTH deficiency		
		Hypopituitarism		
		Hypothalamic-pituitary		
		disease		

Table 6.7 (continued)

Vital signs	Signs and symptoms	Risk factors/ contributing factors	Potential tests/ interventions	Nursing considerations
Hypophysitis [119 123 124] A 57-year-ou	ld male with recurrent meta	stasis renal cell carcinom	a s/n radical left nenhred	comv and right femur
radiation 2 years ago. Current therapy includes nicolumab and ipilimumab. Complicated with immunotherapy related hypophysitis and hypothyraidism requiring high-does steroids, now being tanered down. He presented with new onset dizziness				
BP 104/68	Fatigue	Immunotherapy with	ACTH	Gastric ulcer
HR 126	Headache	ipilimumab	Thyroid panel	prevention
Т 37.2	Dizziness	Hormone imbalances	5 1	High risk for
R 20	Nausea/vomiting			infection
SpO ₂ 95%	Altered mental status			
	Visual disturbances			
	Fever			
Thyroid storm [125–129]. A 59-year-old m	ale with papillary carcinon	na with metastatic disease	e to the cerebellum, cervi	cal nodes, thoracic
nodes, lungs, bones, and spine. The patient	presented with tachycardia	a. The patient was diagnos	sed with thyrotoxicosis w	ith atrial fibrillation.
BP 147/92	Tachycardia	Type 1 diabetes	Thyroid panel	Monitor cardiac
HR 120	Anxiety	Thyroid cancer	Antithyroid	status, at risk for
Т 37.6	Diaphoresis	TSH-secreting pituitary	medication	decreased cardiac
R 22	Atrial fibrillation	adenoma	Cardiac monitoring	output
SpO ₂ 97%	Tremors	Adrenal insufficiency		
		Untreated		
		hyperthyroidism		
Acute ischemic stroke [130–135]. A 73-yea	ar-old female with low-grad	le follicular lymphoma, at	rial fibrillation who pres	ented with tremors and
altered mental status. Patient family stated	approximately 1 hour ago,	the patient started comple	aining of a headache and	l mentation began to
deteriorate. The patient's daughter stated t	hey brought her in when "s	he was not making sense	when she was talking."	
BP 166/82	Altered mental status	Hypertension	Stroke protocol	Ensure patient is
HR 76	Sudden headache	Diabetes	Verify platelet	eligible for tPA prior
Т 37.1	Numbness	Malignant tumor	statements	to initiation
R 20	Ataxia	Atrial fibrillation		
SpO ₂ 94%	Dysphasia			
Acute myocardial infarction [136–140]. A	51-year-old male cancer p	atient. Current suspicion	of cancer. The patient rec	cently (2 days prior)
had a lymph node biopsy of cervical nodul	es; biopsy results pending. I	The patient reporting to th	e ED with chest and bac	k pain.
BP 160/98	Chest pain/pressure	Hypertension, cardiac	EKG	Acute pain
HR 46	Dyspnea	and pulmonary disease	CBC and chemistry	management
Т 36.7	Diaphoresis	Diabetes	Troponin trends	Tissue perfusion
R 22		Cardiotoxic	Interventional	Activity intolerance
SpO ₂ 92%		medications	radiology	Risk for excess fluid
		Hypertension		volume
		Hyperlipidemia		
Pancytopenia [103, 141–144]. A 42-year-o	ld male with a recent diagn	osis of AML and recent in	iduction chemotherapy ti	reatment. Presents with
shortness of breath, gingival and rectal ble	eding.			
BP 90/60	Shortness of breath	Hematologic cancers	CBC with diff	Risk for infection
HR 116	Pallor	Hepatitis	ABORh	Shortness of breath
T 37.2	Fatigue	Chemotherapy	Blood product	caused by anemia
R 20	Bleeding	Recent chemotherapy	replacement	exacerbation with
SpO ₂ 96%	Tachycardia	Sepsis		activity
		Malignancy		Risk for bleeding,
				high-risk for fall with
				Injury Deplese lower the 1
				Replace lowest blood
				product first to
				related to
				1010100 10

COVID 19: An Enhanced Threat to Cancer Patients

It is believed that patients with comorbid conditions, if infected, are at a heightened risk of manifesting complications associated with the virus [153]. Patients with cancer therefore remain at the forefront of this concern. Based on a recent Chinese cohort, patients with cancer had an increased risk of suffering severe events (intensive care unit admission, assisted ventilation or death) compared to those without cancer (39% vs. 8%, p = 0.0003) [154]. The threat the virus poses to medically compromised and noncompromised populations has therefore prompted extensive operational safety measures.

pancytopenia

Table 6.8 Case study: end of life in advanced-stage cancer

		Risk factors/		
		contributing	Potential tests/	
Vital signs	Signs and symptoms	factors	interventions	Nursing considerations
Vital signs <i>Triple threat</i> [145–147]. <i>An 89-year-o</i> <i>breath, altered mental status, and incre</i> <i>caregiver and provides the history. He</i> <i>hospital.</i> BP 101/56 HR 113 T 37.2 R 24 SpO ₂ 92% (on 3 liters/min via nasal cannula)	Signs and symptoms Id male with stage 4 lun, eased lethargy over the <i>i</i> has visited the ED 3 tim Delirium, altered mental status, or confusion in the last week Shortness of breath or difficulty breathing at rest or on minor exertion, such as toileting	factors g cancer on a clinic last couple of days. tes in the last 2 wee Advanced stage cancer "Triple threat" symptoms at triage presentation Clinical trial, potentially indicating "Hail	interventions al trial treatment regim The patient presents wi ks for similar chief com Discuss goals of care early in the encounter to prevent unwanted ICU admissions or invasive procedures Establish code status and provide realistic expectations of care to patient and family	Nursing considerations en presents to triage with shortness of th his adult son who is his primary plaints resulting in admission to the Provide support to patients and families to the best of your ability Use available resources to help provide patients/families with information. It takes a whole team to successfully have "goals of care" conversation in the ED You do not need to have all the answers as an ED nurse. Consult the
	Spending more than 50% of their time in bed, or poor performance status with decreasing independence	Mary" attempt to cure cancer Multiple ED visits may increase in the last weeks of life Patient may be nearing end of life Does the patient have advance care planning documents in place?	Ensure holistic approach and provide all necessary parties to produce a positive outcome	experts in their specific areas (chaplaincy, social work, case management, palliative medicine, hospice, etc.) to assist in planning for those specific needs Ensure advance care planning documentation is available for all to see in the medical record to minimize confusion in an acute event regarding what the patient's wishes truly are Ensure the patient is comfortable – just because you are not providing curative treatment for the cancer does not mean they shouldn't receive treatment for infections, reversible conditions, and symptom management

Preventing Cancer Patients from COVID-19 Exposure from ED to Disposition

Notably, public safety measures in place are designed to reduce preventable hospital admissions and elective procedures [155]. These measures do not fully serve the interests of patients with cancer, who require continuous care inclusive of, but not limited to, diagnostic tests and therapeutic interventions. In this sense, both limitations in medical care and potential COVID-19 exposure could be risky, or even fatal [156].

It is for these reasons that remarkable efforts are taken by hospital personnel to screen for exposure to COVID 19 at hospital entry points. The oncologic ED is a main entry point for patients with cancer and as such, it adheres to the guidelines and recommendations put forth by the Center for Disease and Control (CDC) [157]. We share our adapted screening and preventative measures below.

Screening for COVID-19 and Safety Measures

• Staff member(s) are stationed near all ED and facility entrances (outdoors if weather and facility layout permit), or

in the waiting room area, to ensure patients are screened for symptoms and fever before entering the treatment floor.

- Patients are provided with a face mask upon ED entry.
- Patients are screened for fever or symptoms consistent with COVID-19.
- Patients are directed to designated waiting areas which are divided to separate symptomatic from asymptomatic.
- Patients are separated by at least 6 feet; the area for symptomatic patients is at least 6 feet away from the area for patients without symptoms.
- For patients in need of urgent care, ED providers are notified immediately.
- Alerts and signs are posted in strategic places around the ED and the facility at large, with instructions for patients with fever or symptoms of respiratory infection.

Considerations for ED Staff

 Staff members in charge of screening patients remain 6 feet away from the patient until he or she is determined to be symptom-free and afebrile (temperature is determined by active temperature monitoring).

- Screening staff wear facemasks and shields (for source control) but do not need to wear PPE if they are separated from patients by a physical barrier such as glass or plastic window.
- Screening staff ensures these interactions as brief as possible by limiting the interaction to screening questions only.
- For staff members who must be within 6 feet of a patient, they are required to wear appropriate PPE, including an N95 or higher level respirator, gloves, and eye protection.

Post Patient Screening and Treatment Room Assignment

- Notification of direct patient care staff of the presence of a symptomatic patient.
- Safe and prompt transfer of symptomatic patients from triage to treatment rooms.
- Posting of appropriate isolation signs outside treatment rooms to communicate status.
- Immediate disinfection of waiting areas occupied by symptomatic/exposed patients and surfaces that were within 6 feet of the symptomatic patient; this is in addition to the regular (frequent) baseline cleaning and disinfection process that occurs for the entire waiting area.
- Items that cannot be disinfected remain with the patient or discarded.

Regulatory Standards for Oncologic Emergency Departments: Brief Introduction

Healthcare organizations that achieve accreditation through a Det Norske Veritas (DNV) or The Joint Commission (TJC) "deemed status" survey are determined to meet Medicare and Medicaid requirements and may receive payment from the Center for Medicare and Medicaid Services (CMS). Accreditation does not protect a hospital from an additional CMS survey. All healthcare organizations are still subject to a CMS survey based on a complaint or a validation survey [158]. Validation surveys usually occur within 60 days of the accreditation survey; however, TJC, in collaboration with CMS, has been working on redesigning the validation survey process. The objective of the redesign is to eliminate the validation survey and for CMS to oversee the accreditation process; thus, both may survey an organization at the same time [159].

CMS developed comprehensive Conditions of Participation (CoPs) and Condition for Coverage (CfC) that hospitals and other healthcare entities must meet to initiate or continue their participation in the Medicare and Medicaid programs [158]. All hospitals, including acute care, critical access, long-term care, children's, psychiatric, and cancer hospitals, are included. There are various key conditions of participation chapters for hospitals, and they all involve Emergency Services to varying degrees. Table 6.9 lists CMS subpart chapters applicable to an emergency setting but is

Emergency services, one of the optional services that may be reviewed by CMS, are often an integral part of most hospital surveys. Thus, any organization with an ED will need to adhere to these standards. Table 6.10 displays additional subchapters that each hospital must examine to ensure compliance although some areas may not apply [158].

not all-inclusive list of the regulatory standards [158].

An emergency preparedness plan is required by all healthcare facilities. These guidelines will ensure compliance and demand a proactive approach to adequately plan for natural and man-made disasters. The CMS State Operations Manual, Appendix Z, Emergency Preparedness will guide the development of a comprehensive plan and will likely involve collaboration between the ED and the organization to meet the expectations or standards [160].

CMS is the single largest payer for healthcare in the United States, and the CoP health and safety standards are the foundation or *minimum* standards for its beneficiaries. There are other federal laws that all oncologic urgent or EDs or centers must follow. They include but are not limited to the Emergency Medical Treatment and Labor Act (EMTALA), originally part of the Consolidated Omnibus Budget Reconciliation Act (COBRA) passed in 1986 to address anti-dumping issues. Although motivated by the highly publicized anti-dumping incidents, EMTALA was intended to prevent inadequate care and delay or denial of

Table 6.9 Center for Medicare & Medicaid Services subpart chapters

482.11 Administration	482.25 Pharmaceutical
482.12 Governing body	services
482.13 Patients' rights	482.26 Radiologic services
482.15 Emergency preparedness	482.27 Laboratory services
482.21 Quality assessment and	482.28 Food and dietetic
performance improvement program	services
482.22 Medical staff	482.30 Utilization review
482.23 Nursing services	482.41 Physical
482.24 Medical record services	environment
	482.42 Infection control
	482.43 Discharge planning
	482.45 Organ, tissue & eye
	procurement

 Table 6.10
 Center for Medicare & Medicaid Services optional hospital services chapters

482.51 Surgical services	482.56 Rehabilitation services
482.52 Anesthesia services	482.57 Respiratory care services
482.53 Nuclear medicine services	482.54 Outpatient services
482.55 Emergency services	

EMTALA violations are reported to:	Purpose
Office of Inspector General (OIG)	To issue and enforce civil monetary penalties
Office of Civil Rights	To evaluate if there are any civil rights violations
Justice Department	To evaluate for Hill-Burton Act violations
Internal Revenue Service	To evaluate of tax-exempt status
Joint Commission or Det Norske Veritas (DNV)	To review accreditation status, patterns and trends

 Table 6.11
 Reporting violations of the Emergency Medical Treatment and Labor Act (EMTALA)

treatment of an emergent condition for the uninsured person to include pregnant women seeking medical advice. EMTALA is a federally mandated social policy calling for access to healthcare that hospitals and physicians must address [160, 161]. EMTALA violations are also reported to other regulatory entities listed in Table 6.11. Of note, the most common violation is an inappropriate Medical Screening Examination [161].

The CMS State Operations Manual interpretive guidelines Appendix V is devoted to Emergency Services and EMTALA and provides direction with the EMTALA demands [162]. CMS is responsible for all investigations of EMTALA violations and is partially responsible for enforcements through citations, often designated as "Notice of Termination from Medicare," which gives a hospital 23 days to come into compliance. A plan of correction will need to be submitted with credible evidence of compliance beyond the date of reinspection. On day 19, a notice of termination is published in local newspapers, unless a plan has been submitted, accepted, and re-survey shows compliance within the 23 days [163].

Possible EMTALA violations need to be reported by the receiving hospital within 72 hours, and healthcare organizations have significant sanctions for failure to report, to include termination from Medicare participation. Some states require any healthcare employee with knowledge of a violation to report timely. CMS expects organizations to self-report violations. However, organizational practices vary. Blatant violations may go uncited, while minor or even marginal incongruities may receive punitive enforcement. Often, this variability is related to the interpretation of the law. The Government Accounting Office has reported the variability to Congress, calling for improved consistency. Currently, the inconsistencies continue [162].

Any EMTALA or CMS investigation or validation survey is very demanding for most hospitals. Every detail of hospital operation is often under intense scrutiny. The evaluation of compliance is very black and white, and there is no gray. Either you are compliant, or you are not. Also note, there are no pre-termination appeal rights under EMTALA [158]. COP investigations often lead to "lengthy citations for every dirt mark or dust covering found on any location in the facility. Inspectors are reported to literally surveyed facilities using magnifying glasses and flashlights" [163]. Oncologic hospital administration or nurse leaders do not expect CMS or EMTALA surveys to be as concrete as they are known to be and may struggle significantly. Even minutes are reviewed in detail. There is no gray, only black and white, when determining compliance during a CMS survey. Again, either you are compliant, or you are not [163].

There are two types of citations that CMS can issue. The "condition-level" is considered more serious and indicates that a hospital is not in substantial compliance. A "standard-level" deficiency is cited when a hospital is out of compliance with one aspect of the regulations and it is considered less severe than the condition-level citation. Most surveys have a mix of both types once the final report is released. The hospital has only 10 days to submit a correction plan once they receive the Form CMS-2567 report. If the plan of correction is not accepted as written, the hospital is asked to submit a revised plan [158].

When surveyors determine that the hospital's noncompliance from regulatory standards constitutes an immediate threat to patients' health and safety, they will issue an "Immediate Jeopardy" (IJ) [164]. An IJ determination forces a hospital to immediately stop and correct the underlying problems and is considered the most serious type of violation. Once a hospital or healthcare organization receives an IJ citation, it is given a short time frame to fix the deficiency. If the organization fails to address the IJ as CMS demands, CMS will terminate the facility's Medicare and Medicaid funding. Losing accreditation has a significant impact and can be devastating since the government is the largest payer, and loss of accreditation will affect hospital insurance rates, among other things. It may erode a hospital's infrastructure quickly; physicians stop sending patients, the staff starts leaving, and an organization quickly spirals downward [164].

Over the last few years, there have been several oncologic hospitals that have been surveyed by CMS. The plans of correction are considered public knowledge and are available for review online. They are an excellent source of information to strengthen your organization. Hospitals grow significantly after a survey, becoming stronger and more focused.

Interdisciplinary Collaboration

An essential aspect of providing high-quality care to cancer patients in the ED is the interdisciplinary team's collaboration and cohesion. As displayed in the case studies in this chapter, the cancer patient requires many different needs when presenting to the ED and will encounter many different teams. The collective plan must be centralized around the patient and family and closed-loop communication is vital to preventing errors and for the administration of appropriate treatment.

The cancer patient population's needs require multidisciplinary care to address all aspects and provide holistic and comprehensive care. Communication between teams is essential for preventing errors and identifying issues in the plan of care. High-reliability organizations promote a just culture environment, seeking to improve systems and prevent human error. This means facilitating an environment where every healthcare team member feels supported to identify patient safety issues and speak up when advocating for the patient and family [164].

Recommendations to promote interdisciplinary collaboration include discussions from all stakeholders with practice changes and an opportunity to provide input, professional practice recognition from interdisciplinary members, and establish clear policies and procedures that clearly and concisely delineate role responsibilities. Another great tool for enhancing teamwork is interdisciplinary high-fidelity simulation exercises [165]. This can reveal strengths and opportunities for improvement without patient safety being jeopardized.

This textbook is an excellent example of interdisciplinary collaboration in action. The information can help physicians work more effectively with their nursing partners by providing information relevant to their scope of practice and how it applies to oncologic emergencies while adhering to the regulatory requirements. As the field of oncologic emergencies continues to evolve, the integration of multidisciplinary teams must continue to develop cohesively to create a useful model for patient-centered care.

References

- American Nurses Association. Emergency nursing: a specialty unlike any other. 2020. https://www.myamericannurse.com/ emergency-nursing-a-specialty-unlike-any-other/. 2011. Accessed 19 Sept 2020.
- 2. Emergency Nurses Association. About Emergency Nurses Association. 2020. https://www.ena.org/about. Updated 2020. Accessed 21 Sept 2020.
- Emergency Nurses Association. Emergency nursing is it right for you? 2020. https://www.ena.org/membership/why-emergencynursing. Updated 2020. Accessed 21 Sept 2020.
- Siegel RL, Miller KD, Jemal A. Cancer statistics, 2020. CA Cancer J Clin. 2020;70(1):7–30.
- Gilboy N, Tanabe P, Travers DA, Rosenau AM, Eitel DR. Emergency Severity Index, version 4: implementation handbook. AHRQ publication no. 05-0046-2. Agency for Healthcare Research and Quality: Rockville; 2005.
- McHugh M, Tanabe P, McClelland M, Khare RK. More patients are triaged using the Emergency Severity Index than any other triage acuity system in the United States. Acad Emerg Med. 2012;19(1):106–9.

- Emergency Severity Index (ESI): a triage tool for emergency departments version 4. Content last reviewed May 2020. Rockville: Agency for Healthcare Research and Quality (AHRQ). https://www.ahrq.gov/professionals/systems/hospital/esi/index. html.
- Kosits LM, Jones K. Interruptions experienced by registered nurses working in the emergency department. J Emerg Nurs. 2011;37(1):3–8.
- Duffield C, Conlon L, Kelly M, Catling-Paull C, Stasa H. The emergency department nursing workforce: local solutions for local issues. Int Emerg Nurs. 2010;184(4):181–7.
- Bowman BT. Electrolyte disorders associated with cancer. J Onco-Nephrol. 2017;1(1):30–5.
- Luzum M, Sebolt J, Chopra V. Catheter-associated urinary tract infection, *Clostridioides difficile* colitis, central line–associated bloodstream infection, and methicillin-resistant *Staphylococcus aureus*. Med Clin. 2020;104(4):663–79.
- Levy MM, Evans LE, Rhodes A. The surviving sepsis campaign bundle: 2018 update. Intensive Care Med. 2018;44(6):925–8.
- Alsirafy SA, Raheem AA, Al-Zahrani AS, Mohammed AA, Sherisher MA, El-Kashif AT, Ghanem HM. Emergency department visits at the end of life of patients with terminal cancer: pattern, causes, and avoidability. Am J Hosp Palliat Med. 2016;33(7):658–62.
- 14. Verhoef MJ, de Nijs EJ, Ootjers CS, Fiocco M, Fogteloo AJ, Heringhaus C, et al. End-of-life trajectories of patients with hematological malignancies and patients with advanced solid tumors visiting the emergency department: the need for a proactive integrated care approach. Am J Hosp Palliat Med. 2020;37(9):692–700.
- Patel M, Swofford B, Distler E. Myocardial bridge: bridging the differential diagnosis. Case Rep. 2017;2017:bcr-2017.
- Jain D, Russell RR, Schwartz RG, Panjrath GS, Aronow W. Cardiac complications of cancer therapy: pathophysiology, identification, prevention, treatment, and future directions. Curr Cardiol Rep. 2017;19(5):36.
- Pastores SM, Voigt LP. Acute respiratory failure in the patient with cancer: diagnostic and management strategies. Crit Care Clin. 2010;26(1):21–40.
- Azoulay E, Alberti C, Bornstain C, Leleu G, Moreau D, Recher C, et al. Improved survival in cancer patients requiring mechanical ventilatory support: impact of noninvasive mechanical ventilatory support. Crit Care Med. 2001;29(3):519–25.
- Vencevicius V, Cicenas S. Spontaneous pneumothorax as a first sign of pulmonary carcinoma. World J Surg Oncol. 2009;7:57.
- Steinhäuslin CA, Cuttat JF. Spontaneous pneumothorax. A complication of lung cancer? Chest. 1985;88(5):709–13.
- Yeung KY, Bonnet JD. Bronchogenic carcinoma presenting as spontaneous pneumothorax: case reports with review of literature. Cancer. 1977;39(5):2286–9.
- Minami H, Sakai S, Watanabe A, Shimokata K. Check-valve mechanism as a cause of bilateral spontaneous pneumothorax complicating bronchioloalveolar cell carcinoma. Chest. 1991;100(3):853–5.
- Nishioka M, Fukuoka M, Nakagawa K, Matsui K, Nakajima T. Spontaneous pneumothorax following partial resolution of total bronchial obstruction. Chest. 1993;104(1):160–3.
- Imazio M, Adler Y. Management of pericardial effusion. Eur Heart J. 2013;34(16):1186–97.
- Bodson L, Bouferrache K, Vieillard-Baron A. Cardiac tamponade. Curr Opin Crit Care. 2011;17(5):416–24.
- Johnstone C, Rich SE. Bleeding in cancer patients and its treatment: a review. Ann Palliat Med. 2018;7(2):265–73.
- Yeung SC, Qdaisat A, Chaftari P, Lipe D, Merlin J, Rajha E, et al. Diagnosis and management of immune-related adverse effects of

immune checkpoint therapy in the emergency department. JACEP Open. 2020:1–23.

- Paraschiv B, Dediu G, Iancu A, Bratu O, Diaconu C. Superior vena cava syndrome. Arch Balkan Med Union. 2017;52(1):39–43.
- Meriggi F. Malignant pleural effusion: still a long way to go. Rev Recent Clin Trials. 2019;14(1):24–30.
- Rossel A, Robert-Ebadi H, Combescure C, Grosgurin O, Stirnemann J, Addeo A, et al. Anticoagulant therapy for acute venous thrombo-embolism in cancer patients: a systematic review and network meta-analysis. PLoS One. 2019;14(3):e0213940.
- 31. Elsayem AF, Merriman KW, Gonzalez CE, Yeung SC, Chaftari PS, Reyes-Gibby C, Todd KH. Presenting symptoms in the emergency department as predictors of intensive care unit admissions and hospital mortality in a comprehensive cancer center. J Oncol Pract. 2016;12(5):e554–63.
- 32. El Majzoub I, Abunafeesa H, Cheaito R, Cheaito MA, Elsayem AF. Management of altered mental status and delirium in cancer patients. Ann Palliat Med. 2019;8(5):728–39.
- Dobra M, Bordi L, Nyulas T, Stănescu A, Morariu M, Condrea S, Benedek T. Clinical update. Computed tomography—an emerging tool for triple rule-out in the emergency department. A review. J Cardiovasc Emerg. 2017;3(1):36–40.
- Nolan C, DeAngelis LM. The confused oncologic patient: a rational clinical approach. Curr Opin Neurol. 2016;29(6):789–96.
- 35. Fiordoliva I, Meletani T, Baleani MG, Rinaldi S, Savini A, Di Pietro PM, Berardi R. Managing hyponatremia in lung cancer: latest evidence and clinical implications. Ther Adv Med Oncol. 2017;9(11):711–9.
- Patterson JH. The impact of hyponatremia. Pharmacotherapy. 2011;31(5 Part 2):5S–8S.
- Hermes A, Waschki B, Reck M. Hyponatremia as prognostic factor in small cell lung cancer–a retrospective single institution analysis. Respir Med. 2012;106(6):900–4.
- Hufschmidt A, Shabarin V. Diagnostic yield of cerebral imaging in patients with acute confusion. Acta Neurol Scand. 2008;118(4):245–50.
- Patel PV, Quraishi SA. Management of fluids, electrolytes, and blood products in neurosurgical patients. In: Brambrink A, Kirsch J, editors. Essentials of neurosurgical anesthesia & critical care. Cham: Springer; 2020. p. 31–4.
- 40. Tasler T, Bruce SD. Hyponatremia and SIADH: a case study for nursing consideration. Clin J Oncol Nurs. 2018;22(1):17–9.
- Dabaja BS, Suki D, Pro B, Bonnen M, Ajani J. Adenocarcinoma of the small bowel: presentation, prognostic factors, and outcome of 217 patients. Cancer. 2004;101(03):51826.
- Tchan M. Hyperammonemia and lactic acidosis in adults: differential diagnoses with a focus on inborn errors of metabolism. Rev Endocr Metab Disord. 2018;19(1):69–79.
- 43. Tokuyama S, Fukunaga M, Konishi K, Honda S, Yukimoto R, Okamoto A, et al. A case of hyperammonemia induced by chemotherapy with 5-fluorouracil for metastatic colon cancer. Gan To Kagaku Ryoho. 2018;45(4):743–745. [Article in Japanese].
- 44. Leo C, Wang Y, Mold A, Quintana J, Shi H, Abdullah M, Alaie D, Petrillo R. Noncirrhotic hyperammonemia: a factor behind dementia to alter mental status. Clin Case Rep. 2019;7(11):2118–22.
- Hung TY, Chen CC, Wang TL, Su CF, Wang RF. Transient hyperammonemia in seizures: a prospective study. Epilepsia. 2011;52(11):2043–9.
- Sands JM, Daly ME, Lee EQ. Neurologic complications of lung cancer. Cancer. 2020;126:4455–65. https://doi.org/10.1002/ cncr.32772.
- Selvik HA, Naess H, Kvistad CE. Intravenous thrombolysis in ischemic stroke patients with active cancer. Front Neurol. 2018;9:811.
- Boucher JE, Carpenter D. Sepsis: symptoms, assessment, diagnosis, and the hour-1 bundle in patients with cancer. Clin J Oncol Nurs. 2020;24(1):99–102.

- Ruff MW, Porter AB. Neuro-oncologic emergencies. In: Rabinstein A, editor. Neurological emergencies. Cham: Springer; 2020. p. 107–19.
- 50. Schneider GH, von Helden A, Franke R, Lanksch WR, Unterberg A. Influence of body position on jugular venous oxygen saturation, intracranial pressure and cerebral perfusion pressure. In: Unterberg AW, Schneider GH, Lanksch WR, editors. Monitoring of cerebral blood flow and metabolism in intensive care. Acta Neurochirurgica, vol. 59. Vienna: Springer; 1993. p. 107–12.
- Schiff D, O'Neill BP, Suman VJ. Spinal epidural metastasis as the initial manifestation of malignancy: clinical features and diagnostic approach. Neurology. 1997;49(2):452–6.
- Al-Qurainy R, Collis E. Metastatic spinal cord compression: diagnosis and management. BMJ. 2016;353:i2539.
- Maranzano E, Latini P, Beneventi S, Perruci E, Panizza BM, Aristei C, et al. Radiotherapy without steroids in selected metastatic spinal cord compression patients. A phase II trial. Am J Clin Oncol. 1996;19(2):179–83.
- Kaplow R, Iyere K. Understanding spinal cord compression. Nursing. 2016;46(9):44–51.
- Husband DJ, Grant KA, Romaniuk CS. MRI in the diagnosis and treatment of suspected malignant spinal cord compression. Br J Radiol. 2001;74(877):15–23.
- Kumar A, Weber MH, Gokaslan Z, Wolinsky JP, Schmidt M, Rhines L, et al. Metastatic spinal cord compression and steroid treatment. Clin Spine Surg. 2017;30(4):156–63.
- Boussios ST, Cooke D, Hayward C, Kanellos FS, Tsiouris AK, Chatziantoniou AA, et al. Metastatic spinal cord compression: unraveling the diagnostic and therapeutic challenges. Anticancer Res. 2018;38(9):4987–97.
- Catena F, De Simone B, Coccolini F, Di Saverio S, Sartelli M, Ansaloni L. Bowel obstruction: a narrative review for all physicians. World J Emerg Surg. 2019;14:20.
- Ferguson HJ, Ferguson CI, Speakman J, Ismail T. Management of intestinal obstruction in advanced malignancy. Ann Med Surg (Lond). 2015;4(3):264–70.
- Pujahari AK. Decision making in bowel obstruction: a review. J Clin Diagn Res. 2016;10(11):PE07–12.
- Tabchouri N, Dussart D, Giger-Pabst U, Michot N, Marques F, Khalfallah M, et al. Only surgical treatment to be considered for adhesive small bowel obstruction: a new paradigm. Gastroenterol Res Pract. 2018;2018:9628490.
- Singh P, Yoon SS, Kuo B. Nausea: a review of pathophysiology and therapeutics. Ther Adv Gastroenterol. 2016;9(1):98–112.
- Mith DA, Kashyap S, Nehring SM. Bowel obstruction. [Updated 2020 Aug 10]. In: StatPearls [Internet]. Treasure Island: StatPearls Publishing; 2020. Available from: https://www.ncbi.nlm.nih.gov/ books/NBK441975/.
- Balaban CD, Yates BJ. What is nausea? A historical analysis of changing views. Auton Neurosci. 2017;202:5–17.
- Bowen JM. Mechanisms of TKI-induced diarrhea in cancer patients. Curr Opin Support Palliat Care. 2013;7(2):162–7.
- 66. Sweetser S. Evaluating the patient with diarrhea: a case-based approach. Mayo Clin Proc. 2012;87(6):596–602.
- Forootan M, Bagheri N, Darvishi M. Chronic constipation: a review of literature. Medicine (Baltimore). 2018;97(20):e10631.
- Webster LR. Opioid-induced constipation. Pain Med. 2015;16(Suppl 1):S16–21.
- Navari RM. Managing nausea and vomiting in patients with cancer: what works. Oncology (Williston Park). 2018;32(3):121–5, 131, 136.
- 70. Riess H, Ay C, Bauersachs R, Becattini C, Beyer-Westendorf J, Cajfinger F, et al. Use of direct oral anticoagulants in patients with cancer: practical considerations for the management of patients with nausea or vomiting. Oncologist. 2018;23(7):822–39.
- 71. Schwartzberg LS, Rugo HS, Aapro MS. New and emerging therapeutic options for the management of chemotherapy-induced

nausea and vomiting. Clin Adv Hematol Oncol. 2015;13(3 Suppl 3):3–13.

- Serlin DC, Heidelbaugh JJ, Stoffel JT. Urinary retention in adults: evaluation and initial management. Am Fam Physician. 2018;98(8):496–503.
- Gray M. Urinary retention. Management in the acute care setting. Part. 2. Am J Nurs. 2000;100(8):36–43; quiz 44.
- Selius BA, Subedi R. Urinary retention in adults: diagnosis and initial management. Am Fam Physician. 2008;77(5):643–50.
- Billet M, Windsor TA. Urinary retention. Emerg Med Clin North Am. 2019;37(4):649–60.
- Willette PA, Coffield S. Current trends in the management of difficult urinary catheterizations. West J Emerg Med. 2012;13(6):472–8.
- Campbell GA, Hu D, Okusa MD. Acute kidney injury in the cancer patient. Adv Chronic Kidney Dis. 2014;21(1):64–71.
- Yarmohammadi H, Covey AM. Percutaneous biliary interventions and complications in malignant bile duct obstruction. Chin Clin Oncol. 2016;5(5):68.
- Coucke EM, Akbar H, Kahloon A, Lopez PP. Biliary obstruction. [Updated 2020 Jun 22]. In: StatPearls [Internet]. Treasure Island: StatPearls Publishing; 2020. Available from: https://www.ncbi. nlm.nih.gov/books/NBK539698/.
- Tsalis K, Zacharakis E, Vasiliadis K, Kalfadis S, Vergos O, Christoforidis E, Betsis D. Bile duct injuries during laparoscopic cholecystectomy: management and outcome. Am Surg. 2005;71(12):1060–5.
- Morani AC, Hanafy AK, Marcal LP, Subbiah V, Le O, Bathala TK, Elsayes KM. Imaging of acute abdomen in cancer patients. Abdom Radiol. 2020;45(8):2287–304.
- Salihah N, Mazlan N, Lua PL. Chemotherapy-induced nausea and vomiting: exploring patients' subjective experience. J Multidiscip Health. 2016;9:145–51.
- Meehan CD, McKenna C. Preventing hospital-acquired pneumonia. Am Nurse J. 2020;15(2):16–21. https://www. myamericannurse.com/wp-content/uploads/2020/02/an2-CE-Pneumonia-309a.pdf.
- 84. Hammami MB, Talkin R, Al-Taee AM, Schoen MW, Goyal SD, Lai JP. Autologous graft-versus-host disease of the gastrointestinal tract in patients with multiple myeloma and hematopoietic stem cell transplantation. Gastroenterol Res. 2018;11(1):52–7.
- Stukan M. Drainage of malignant ascites: patient selection and perspectives. Cancer Manage Res. 2017;9:115–30.
- Bell T, O'Grady NP. Prevention of central line–associated bloodstream infections. Infect Dis Clin. 2017;31(3):551–9.
- Tebano G, Mouelhi Y, Zanichelli V, Charmillon A, Fougnot S, Lozniewski A, et al. Selective reporting of antibiotic susceptibility testing results: a promising antibiotic stewardship tool. Expert Rev Anti Infect Ther. 2020;18(3):251–62.
- Pasikhova Y, Ludlow S, Baluch A. Fever in patients with cancer. Cancer Control. 2017;24(2):193–7.
- Zimmer AJ, Freifeld AG. Optimal management of neutropenic fever in patients with cancer. J Oncol Pract. 2019;15(1):19–24.
- Williams MD, Braun LA, Cooper LM, Johnston J, Weiss RV, Qualy RL, Linde-Zwirble W. Hospitalized cancer patients with severe sepsis: analysis of incidence, mortality, and associated costs of care. Crit Care. 2004;8(5):R291–8.
- 91. Del Principe MI, Buccisano F, Maurillo L, Venditti D, Cefalo M, Sarlo C, et al. Infections increase the risk of central venous catheter-related thrombosis in adult acute myeloid leukemia. Thromb Res. 2013;132(5):511–4.
- 92. Bate J, Gibson F, Johnson E, Selwood K, Skinner R, Chisholm J, National Institute for Clinical Excellence. Neutropenic sepsis: prevention and management of neutropenic sepsis in cancer patients (NICE Clinical Guideline CG151). Arch Dis Child Educ Pract Ed. 2013;98(2):73–5.

- Sheats MK. A comparative review of equine SIRS, sepsis, and neutrophils. Front Vet Sci. 2019;6:69.
- Zhou Y, Abel GA, Hamilton W, Pritchard-Jones K, Gross CP, Walter FM, et al. Diagnosis of cancer as an emergency: a critical review of current evidence. Nat Rev Clin Oncol. 2017;14(1):45–56.
- Lam KF, Lim HA, Kua EH, Griva K, Mahendran R. Mindfulness and cancer patients' emotional states: a latent profile analysis among newly diagnosed cancer patients. Mindfulness. 2018;9(2):521–33.
- Livingood WC, Smotherman C, Lukens-Bull K, Aldridge P, Kraemer DF, Wood DL, Volpe C. An elephant in the emergency department: symptom of disparities in cancer care. Popul Health Manag. 2016;19(2):95–101.
- Blackburn LM, Bender S, Brown S. Acute leukemia: diagnosis and treatment. Semin Oncol Nurs. 2019;35(6):150950.
- Shallis RM, Wang R, Davidoff A, Ma X, Zeidan AM. Epidemiology of acute myeloid leukemia: recent progress and enduring challenges. Blood Rev. 2019;36:70–87.
- Lowenberg B, Downing JR, Burnett A. Acute myeloid leukemia. N Engl J Med. 1999;341(14):1051–62.
- 100. Mamez AC, Raffoux E, Chevret S, Lemiale V, Boissel N, Canet E, et al. Pre-treatment with oral hydroxyurea prior to intensive chemotherapy improves early survival of patients with high hyperleukocytosis in acute myeloid leukemia. Leuk Lymphoma. 2016;57(10):2281–8.
- 101. Arber DA, Borowitz MJ, Cessna M, Etzell J, Foucar K, Hasserjian RP, et al. Initial diagnostic workup of acute leukemia: guideline from the College of American Pathologists and the American Society of Hematology. Arch Pathol Lab Med. 2017;141(10):1342–93.
- 102. Montesinos P, Lorenzo I, Martín G, Sanz J, Pérez-Sirvent ML, Martínez D, et al. Tumor lysis syndrome in patients with acute myeloid leukemia: identification of risk factors and development of a predictive model. Haematologica. 2008;93(1):67–74.
- Douer D. The epidemiology of acute promyelocytic leukaemia. Best Pract Res Clin Haematol. 2003;16(3):357–67.
- 104. Bittencourt H, Teixeira Junior AL, Glória AB, Ribeiro AF, Fagundes EM. Acute promyelocytic leukemia presenting as an extradural mass. Rev Bras Hematol Hemoter. 2011;33(6):478–80.
- 105. Wang ZY, Chen Z. Acute promyelocytic leukemia: from highly fatal to highly curable. Blood. 2008;111(5):2505–15.
- 106. Gnanaraj J, Parnes A, Francis CW, Go RS, Takemoto CM, Hashmi SK. Approach to pancytopenia: diagnostic algorithm for clinical hematologists. Blood Rev. 2018;32(5):361–7.
- 107. Anract P, Biau D, Boudou-Rouquette P. Metastatic fractures of long limb bones. Orthop Traumatol Surg Res. 2017;103(1):S41–51.
- 108. Yeung SC. Endocrine and metabolic emergencies. In: Todd KH, Thomas Jr CR, editors. Oncologic emergency medicine. 1st ed. Cham: Springer International; 2016. p. 243–62.
- Nassereddine S, Rafei H, Elbahesh E, Tabbara I. Acute graft versus host disease: a comprehensive review. Anticancer Res. 2017;37(4):1547–55.
- Chang VY, Wang JJ. Pharmacogenetics of chemotherapy-induced cardiotoxicity. Curr Oncol Rep. 2018;20(7):52.
- 111. Liu D, Zhao J. Cytokine release syndrome: grading, modeling, and new therapy. J Hematol Oncol. 2018;11(1):121.
- 112. Flory J, Farooki A. Diabetes management in cancer patients. Oncology (Williston Park). 2016;30(6):565–70.
- 113. Feo R, Conroy T, Marshall RJ, Rasmussen P, Wiechula R, Kitson AL. Using holistic interpretive synthesis to create practicerelevant guidance for person-centered fundamental care delivered by nurses. Nurs Inq. 2017;24(2):e12152.
- Fishman JA, Hogan JI, Maus MV. Inflammatory and infectious syndromes associated with cancer immunotherapies. Clin Infect Dis. 2019;69(6):909–20.

- 115. Damron TA. Update on predicting fracture risk: what's new, what's old, what's accurate. Tech Orthop. 2018;33(3):191–200.
- Riemen AH, Aherne B, Bruce E, Boddie DE, McCullough LA. Pathological hip fractures. Orthop Trauma. 2020;34(2):64–72.
- 117. Rome S, Noonan K, Bertolotti P, Tariman JD, Miceli T. Bone health, pain, and mobility: evidence-based recommendations for patients with multiple myeloma. Clin J Oncol Nurs. 2017;21(5 Suppl):47–59.
- Diabetes Canada Clinical Practice Guidelines Expert Committee, Goguen J, Gilbert J. Hyperglycemic emergencies in adults. Can J Diabetes. 2018;42(Suppl 1):S109–14.
- 119. Higham CE, Chatzimavridou-Grigoriadou V, Fitzgerald CT, Trainer PJ, Eggermont AM, Lorigan P. Adjuvant immunotherapy: the sting in the tail. Eur J Cancer. 2020;132:207–10.
- 120. El Hussein MT, Kilfoil L. Using your "head backward" in the management of diabetic ketoacidosis. Nurse Pract. 2020;45(1):7– 11. https://doi.org/10.1097/01.npr.0000615576.42292.68.
- Rushworth RL, Torpy DJ, Falhammar H. Adrenal crisis. N Engl J Med. 2019;381(9):852–61.
- 122. Feeney C, Buell K, Avari P, Buckley A, Meeran K, Rees D. Addisonian crisis: assessment and management. Br J Hosp Med (Lond). 2018;79(3):C34–7.
- Valecha G, Pant M, Ibrahim U, Atallah JP. Immunotherapyinduced autoimmune hypophysitis. J Oncol Pharm Pract. 2019;25(1):217–20.
- 124. Cooksley T, Knight T, Gupta A, Higham C, Lorigan P, Adam S. Emergency ambulatory outpatient management of immune-mediated hypophysitis. Support Care Cancer. 2020;28(9):3995–9.
- 125. Wallington D, Schauer M, Bauler LD. Simultaneous presentation of thyroid storm and diabetic ketoacidosis in a previously healthy 21-year-old man. BMJ Case Rep. 2019;12(1):bcr-2018-227554.
- 126. Leung AM. Thyroid emergencies. J Infus Nurs. 2016;39(5):281-6.
- Kearney T, Dang C. Diabetic and endocrine emergencies. Postgrad Med J. 2007;83(976):79–86.
- 128. Andrade Luz I, Pereira T, Catorze N. Thyroid storm: a case of haemodynamic failure promptly reversed by aggressive medical therapy with antithyroid agents and steroid pulse. BMJ Case Rep. 2018;11(1):e226669.
- 129. Schreiber ML. Thyroid storm. Medsurg Nurs. 2017;26(2):143-5.
- Park JH, Kim BS, Ovbiagele B. Concomitant diabetes or hypertension with metabolic syndrome on the extent of intracranial atherosclerotic stenosis. Neurol Sci. 2020;41(2):387–95.
- 131. Wang F, Hu XY, Cui ZM, Fang XM, Dai Z, Wang T, Guo DL. Clinical and imaging characteristics of malignant tumor concurrent with stroke. Cancer Biother Radiopharm. 2019;34(8):504–10.
- 132. El Ammar F, Ardelt A, Del Brutto VJ, Loggini A, Bulwa Z, Martinez RC, et al. BE-FAST: a sensitive screening tool to identify in-hospital acute ischemic stroke. J Stroke Cerebrovasc Dis. 2020;29(7):104821.
- 133. Duan Z, Guo W, Tang T, Tao L, Gong K, Zhang X. Relationship between high-sensitivity C-reactive protein and early neurological deterioration in stroke patients with and without atrial fibrillation. Heart Lung. 2020;49(2):193–7.
- 134. Puig J, Shankar J, Liebeskind D, Terceño M, Nael K, Demchuk AM, et al. From "time is brain" to "imaging is brain": a paradigm shift in the management of acute ischemic stroke. J Neuroimaging. 2020;30(5):562–71.
- 135. Marko M, Posekany A, Szabo S, Scharer S, Kiechl S, Knoflach M, et al. Austrian Stroke Unit Registry Collaborators. Trends of r-tPA (recombinant tissue-type plasminogen activator) treatment and treatment-influencing factors in acute ischemic stroke. Stroke. 2020;51(4):1240–7.
- 136. Hochman JS, Tamis JE, Thompson TD, Weaver WD, White HD, Van de Werf F, et al. Sex, clinical presentation, and out-

come in patients with acute coronary syndromes. N Engl J Med. 1999;341(4):226–32.

- 137. Ewer MS, Ewer SM. Cardiotoxicity of anticancer treatments. Nat Rev Cardiol. 2015;12(9):547–58. Erratum in: Nat Rev Cardiol. 2015;12(11):620.
- 138. Lu L, Liu M, Sun R, Zheng Y, Zhang P. Myocardial infarction: symptoms and treatments. Cell Biochem Biophys. 2015;72(3):865–7.
- 139. Haybar H, Pezeshki SM, Saki N. Evaluation of complete blood count parameters in cardiovascular diseases: an early indicator of prognosis? Exp Mol Pathol. 2019;110:104267.
- Anderson JL, Morrow DA. Acute myocardial infarction. N Engl J Med. 2017;376(21):2053–64.
- 141. Weinzierl EP, Arber DA. Bone marrow evaluation in new-onset pancytopenia. Human Pathol. 2013;44(6):1154–64.
- 142. Rauff B, Idrees M, Shah SA, Butt S, Butt AM, Ali L, Hussain A, Irshad-Ur-Rehman, Ali M. Hepatitis associated aplastic anemia: a review. Virol J. 2011;8:87.
- 143. Hayat AS, Khan AH, Baloch GH, Shaikh N. Pancytopenia: study for clinical features and etiological pattern at tertiary care settings in Abbottabad. Prof Med J. 2014;21(1):060–5.
- 144. Connell NT. Transfusion medicine. Prim Care. 2016;43(4):651-9.
- 145. Elsayem AF, Warneke CL, Reyes-Gibby CC, Page VD, Buffardi LJ, Brock PA, et al. Triple threat of mortality among advance cancer patients presenting to an emergency department. J Clin Oncol. 2017;35(31 Suppl):138.
- 146. Bischoff KE, Sudore R, Miao Y, Boscardin WJ, Smith AK. Advance care planning and the quality of end-of-life care in older adults. J Am Geriatr Soc. 2013;61(2):209–14.
- 147. Decker K, Lee S, Morphet J. The experiences of emergency nurses in providing end-of-life care to patients in the emergency department. Australas Emerg Nurs J. 2015;18(2):68–74.
- 148. Dobbins EH. Improving end-of-life care: recommendations from the IOM. Nurse Pract. 2016;41(9):26–34.
- 149. Sutherland R. Focus: death: dying well-informed: the need for better clinical education surrounding facilitating end-of-life conversations. Yale J Biol Med. 2019;92(4):757–64.
- 150. Li Y, Qin JJ, Wang Z, Yu Y, Wen YY, Chen XK, et al. Surgical treatment for esophageal cancer during the outbreak of COVID-19. Zhonghua Zhong Liu Za Zhi. 2020;42(4):296–300. [Article in Chinese].
- 151. Center for Disease Control and Prevention. Coronavirus disease 2019 (COVID-19). About COVID-19. [Internet] Updated 1 Sept 2020. https://www.cdc.gov/coronavirus/2019-ncov/cdcresponse/ about-COVID-19.html. Accessed 4 Oct 2020.
- 152. World Health Organization. Coronarvirus disease 2019 (COVID-19). Situation Report-176. Data as reported by national authorities by 10:00 CET 5 Apr 2020. Accessed 13 Apr 2020.
- 153. Guan WJ, Liang WH, Zhao Y, Liang HR, Chen ZS, Li YM, et al. China Medical Treatment Expert Group for COVID-19. Comorbidity and its impact on 1590 patients with COVID-19 in China: a nationwide analysis. Eur Respir J. 2020;55(5):2000547.
- 154. Liang W, Guan W, Chen R, Wang W, Li J, Xu K, et al. He J. Cancer patients in SARS-CoV-2 infection: a nationwide analysis in China. Lancet Oncol. 2020;21(3):335–7.
- 155. Cao Y, Li Q, Chen J, Guo X, Miao C, Yang H, et al. Hospital emergency management plan during the COVID-19 epidemic. Acad Emerg Med. 2020;27(4):309–11.
- 156. Moujaess E, Kourie HR, Ghosn M. Cancer patients and research during COVID-19 pandemic: a systematic review of current evidence. Crit Rev Oncol Hematol. 2020;150:102972.
- 157. Center for Disease Control and Prevention. Coronavirus disease 2019 (COVID-19). Healthcare workers. Screening and triage at intake. Screening dialysis patients for COVID-19. [Internet] Updated 14 Apr 2020. https://www.cdc.gov/coronavirus/2019ncov/hcp/dialysis/screening.html.

- 158. Center for Medicare & Medicaid Services. State operations manual. Appendix A. Survey protocol, regulations and interpretive guidelines for hospitals. Rev. 200, 02-21-20. https://www.cms. gov/Regulations-and-Guidance/Guidance/Manuals/Downloads/ som107ap_a_hospitals.pdf. Accessed 20 Sept 2020.
- 159. The Joint Commission. Leading hospital improvement. Pelletier M. CMS revising validation survey process. 22 Aug 2019. https:// www.jointcommission.org/resources/news-and-multimedia/ blogs/leading-hospital-improvement/2019/08/cms-revisingvalidation-survey-process/. Accessed 12 Sept 2020.
- 160. Center for Medicare & Medicaid Services. State operations manual. Appendix Z. Emergency preparedness for all provider and certified supplier types. Interpretive guidance. Rev. 200, 02-21-20. https://www.cms.gov/Regulations-and-Guidance/Guidance/ Manuals/downloads/som107ap_z_emergprep.pdf. Accessed 12 Sept 2020.
- Ladd M, Gupta V. Cobra laws and EMTALA. [Updated 2020 Apr 5]. In: StatPearls [Internet]. Treasure Island: StatPearls Publishing; 2020. https://www.ncbi.nlm.nih.gov/books/NBK555935.

- 162. Center for Medicare & Medicaid Services. State operations manual. Appendix V – Interpretive guidelines – Responsibilities of Medicare participating hospitals in emergency cases. Rev. 191, 07-19-19. https://www.cms.gov/Regulations-and-Guidance/ Guidance/Manuals/downloads/som107ap_v_emerg.pdf. Accessed 12 Sept 2020.
- 163. Frew S, Giese K. EMTALA field guide. 4th ed. Loves Park. Independently published; 2019. p. 33.
- 164. Center for Medicare & Medicaid Services. State operations manual. Appendix Q – Core guidelines for determining immediate Jeopardy. Rev. 187, Issued: 03-06-19. https://www.cms. gov/Regulations-and-guidance/Guidance/Manuals/downloads/ som107ap_q_immedjeopardy.pdf. Accessed 12 Sept 2020.
- 165. Bursiek AA, Hopkins MR, Breitkopf DM, Grubbs PL, Joswiak ME, Klipfel JM, Johnson KM. Use of high-fidelity simulation to enhance interdisciplinary collaboration and reduce patient falls. J Patient Saf. 2020;16(3):245–50.

Palliative Social Work

Robin Rudy Lawson and Alison Snow

Introduction

Cancer patients may require emergency department (ED) care as a result of symptoms or complications of the disease itself, from the side effects of cancer treatment, including chemotherapy or radiation therapy, or from intercurrent injury or illness unrelated to cancer. Sepsis, pain, neutropenia, fever, deep vein thrombosis, nausea/vomiting, and failure to thrive are common symptoms that prompt oncologic emergency care [1-3]. Despite the overall decrease in cancer death rates, ED visits for cancer-related emergencies are on the rise [4]. One Australian study showed that while the number of deaths in the ED for cancer and non-cancerrelated presentations was similar, the number of inpatient deaths for patients with cancer was significantly higher than for non-cancer patients [5]. The impact of a new cancer diagnosis, worsening prognosis, or transition to end-oflife care exacts physical, emotional, financial, and psychosocial distress on patients and families. Medical social workers are key interdisciplinary team members within the inpatient, outpatient, and home care settings who intervene along the continuum of illness. They are specifically trained to assess the patient's adjustment to illness and treatment, as well as any social and financial concerns that may impact medical decision-making. Social workers also provide therapeutic interventions to enhance patient coping, reduce caregiver distress, and ensure continuity of care across settings [6].

When patients with oncologic emergencies and their families present to the ED, social workers may be called upon to intervene with issues involving the entire spectrum of cancer, from a new diagnosis to end-of-life care. In both inpatient and outpatient settings, oncologic social workers are trained

R. R. Lawson (⊠) Empath Health, Suncoast Hospice, St. Petersburg, FL, USA

A. Snow Cancer Supportive Services, Cancer Center, Mount Sinai Beth Israel, New York, NY, USA to provide a variety of interventions to assist patients and families coping with cancer, including assessment of psychosocial needs, adjustment to illness and side effects of treatment, and linkage to community resources. Palliative social workers may also intervene to assist oncologic patients in crisis where primary attention may be placed on pain and symptom management, goals of care conversations, advance care planning, or education and counseling regarding end-oflife care. This chapter examines social work's involvement with oncologic patients in the ED, as well as the oncologic social work role in the outpatient setting, and suggests potential partnerships and collaboration among ED, oncologic, and palliative social work.

The Role of the ED Social Worker

The role of the ED social worker varies greatly, given the diversity of the patient population and the emergent nature of many social work referrals in this setting. ED social work consults are imperative for trauma victims and their families, victims of assault, homeless individuals, persons with substance use disorders, and minors [7]. Due to the time-intensive nature of these referrals, it may be difficult for the ED social worker to consult on less urgent referrals, including patients who, after a diagnostic evaluation, confront a newly diagnosed cancer. Despite these constraints, ED social workers may be the first psychosocial clinicians to see patients with life-limiting illnesses who present with distressing symptoms or who may be actively dying [8].

ED social workers strive to maintain a careful balance by blending their responsibility to provide concrete services such as community resource referrals, medical equipment/ home health setup, and placement in short-term or long-term care facilities, with therapeutic interventions, including crisis intervention. ED social workers also conduct psychosocial assessments and provide bereavement counseling when deaths occur in the ED [9]. A 2016 study in a large, urban trauma ED identified that the most common ED social work

© Springer Nature Switzerland AG 2021



K. H. Todd et al. (eds.), Oncologic Emergency Medicine, https://doi.org/10.1007/978-3-030-67123-5_7

services provided were mental health services, followed by care coordination and then community resource information/referral [7].

Crisis Intervention

For ED social workers, priority referrals that require immediate intervention include domestic violence and sexual assault and elder or child abuse and neglect. In addition to providing support and crisis intervention, these referrals may also require consultation with law enforcement or state agencies as well as hospital security, as well as locating and supporting family members or friends, and maintaining constant communication with the ED physician and nurse. Given the sensitive and complicated nature of this work, the ED social worker may spend a great deal of time on such consultations. Crisis intervention skills utilized with victims of sexual assault and domestic violence include re-establishing coping skills, problem-solving to identify next steps, assessment for suicidality, provision of emotional support, and the resolution of concrete needs such as safety, housing, and financial support.

For patients who present with symptoms of depression, anxiety, suicidal ideation, or alcohol and drug abuse, but do not require inpatient admission, the ED social worker typically completes an assessment and provides brief counseling as well as linkage to community resources to ensure followup support. ED social workers with strong backgrounds in mental health may identify appropriate referrals to psychiatry or outpatient mental health agencies and provide brief interventions and education to interdisciplinary team members on strategies to help patients in this setting.

Complex Case Management

When a patient's ability to care for himself or herself is diminished but admission is not required, the ED social worker may be responsible for obtaining support through home-based health services. They also assist with applying for financial assistance to facilitate admission to a skilled nursing facility, as well as coordinating a plan for family and friends to support caregiving at home. For patients who need additional services, preventing a "social admission" by facilitating a safe discharge from the ED is important.

A Canadian study found that the ED social workers commonly assisted patients with financial concerns and discharge planning needs as well as adjusting to their illness and addiction [10]. Effective discharge planning reduces unnecessary hospital admissions and coordinates the outpatient services a patient needs to allow them to safely remain in their home environment. The expertise and range of skills implicit in crisis intervention, mental health assessment, and complex case management are essential to the care of oncologic patients and their families who present to the ED for help.

ED Social Work Role with Oncologic Patients

When oncologic patients present to the ED with complications from their disease or treatment, the ED social worker may be asked to see patients with concomitant symptoms of depression, anxiety, or suicidal ideation, while also evaluating caregivers who may be exhausted from providing care. Newly diagnosed cancer patients, as well as cancer patients who come to the ED in lieu of a visit with their primary physician or oncologist, may be referred to the ED social worker for assistance with medical follow-up, evaluation, and treatment, as well as provision of concrete services such as home health care or medical equipment.

When oncologic patients present to the ED at the end of life, the ED social worker provides emotional support to the patient and family, communicates with the hospital chaplain as requested by the family, and helps identify a private space for family members to gather and grieve. While there is no published research delineating the role of ED social work with oncologic patients, many of the skills outlined above are integral to assisting oncologic patients and their families.

Psychosocial Issues for Cancer Patients in the ED

Diagnosis of Cancer in the ED

A new diagnosis of cancer may elicit strong emotions and can induce a great deal of stress and anxiety for patients and families. It may cause the patient to experience feelings of loneliness, abandonment, and loss of control over their situation [11]. The ED social worker may be called upon to provide emotional support while identifying and clarifying, in consultation with the emergency physician, any real or perceived fears surrounding a new diagnosis, prognosis, or treatment. Providing patients and families with the necessary time to integrate a new diagnosis is essential to allowing them to consider decisions about appropriate treatment and continuing medical care [11]. For patients who decline further evaluation and treatment, it is essential that the patients understand the potential consequences of avoiding further care. An ED social work assessment of concerns and fears following a new diagnosis of cancer may provide the patient with an alternative plan of care aside from hospitalization,

such as follow-up with an oncologist to review and integrate medical options. Below is a case study of a newly diagnosed cancer patient in the ED that demonstrates the ED social worker's role with patients/families and the interdisciplinary team.

Case Discussion 1

Emergency Department Social Worker

The ED social worker receives a call from an emergency physician to see a 63-year-old Latina woman, who, with her three adult children, is anxiously awaiting the results of a CT scan of her pelvis and abdomen. Prior to this ED visit, her medical history includes diabetes and asthma. She presents with rectal bleeding, severe abdominal pain, and dizziness. The emergency physician informs the ED social worker that the patient has a new diagnosis of metastatic anal cancer and that he would like the social worker to assist as he provides this information to the patient and her family. The patient has no primary care physician and must be admitted to the hospital for further evaluation and consideration of treatment options.

The ED social worker had spoken with the patient earlier in the day while the patient was waiting. The patient confided, "I bleed every time I'm on the toilet, for the last few months." When the social worker inquires if she had told anyone, the patient states that she did not want to tell her family because she was scared. As the social worker begins to explore the patient's fears about telling family, the patient begins crying and pulls out rosary beads from her pocket. The patient explains that her youngest daughter is getting married in a few months and that the focus should be on the young, not the old. She goes on to say she has led a full and happy life and that it is up to God to decide her fate.

With the patient's permission, the ED social worker contacts the Catholic chaplain on call for the ED to be present at the family meeting. The ED social worker secures a private space for the patient and family to meet so that they can process this new diagnosis in a quiet setting. As the emergency physician explains the results of the tests done in the ED and describes the patient's diagnosis, prognosis, and treatment options, the patient and her children begin to cry. While the social worker and the chaplain comfort the patient/ family, the emergency physician sits quietly, in order to allow them time to process this information.

After a few minutes, the emergency physician confirms with the patient that she has heard the words and is beginning to integrate the significance of her medical condition and the treatment options. Following this discussion, the ED social worker acknowledges the unique emotional responses of the patient and family and describes the oncologic and/or palliative services that can be provided concurrently with the chemotherapy or radiation that the patient may receive while in the hospital. The emergency physician concurs with the ED social worker, describing how the patient may benefit from specialized symptom management and the provision of psychosocial and spiritual services for both the patient and family. The patient is admitted to the hospital late that evening and is seen by oncologic and palliative consultants the next day. The ED social worker ensures a seamless transition by communicating with the inpatient unit social worker.

Communication in the ED

Due to an often chaotic and busy environment, ED clinicians are often unable to spend significant periods of time with patients/families. As most patients arrive to the ED in crisis and distress, it may be difficult for them to comprehend a complete picture of their medical problem and proposed treatment. The medical jargon and complex terms used by healthcare professionals may represent a foreign language to patients/families, regardless of their educational level. ED social workers can assess the patient's/family's health literacy and understanding of medical information and then work with the physician to clarify that the patient understands the care they are receiving. The essential role that social workers play as part of the healthcare system is evidenced by their frequent initiation, implementation, and support of end-oflife discussions with patients, families, and other caregivers [12]. Furthermore, social workers advocate on behalf of patients to physicians and nurses, increasing awareness of psychosocial issues along the continuum of illness and the benefits of early referrals to palliative care or hospice.

In addition to normalizing the challenge of integrating information when in crisis, the care taken to help patients and families to anticipate the next steps in resolving their medical crisis models a relationship that highlights both the emotional and informational needs of patients and families. Despite the often limited time that is spent with a patient, all clinicians who care for seriously ill patients contribute to the meaning making and processing of coping with illness. While ED interactions may be brief, they can forever impact patients and families.

ED as Primary Healthcare Site

Patients who typically use the ED as their primary healthcare site are more likely to learn of their cancer diagnosis in the ED. The fragmentary nature of emergency care may limit the possibility of establishing continuity of care and a predictable relationship with a provider after diagnosis [13]. For

R. R. Lawson and A. Snow

patients with advanced cancer who present to the emergency department, one British study found that anxiety related to the disease related to patient interpretation of symptoms, feelings of comfort and safety within the hospital environment, difficulties accessing community healthcare services especially after hours or for urgent reasons, and previous patterns of health-seeking behavior are the major drivers of seeking care in this setting [14].

For those individuals without insurance, social work involvement can be pivotal to helping patients organize their medical care and access available financial assistance, without which, treatment for their disease can be delayed and suboptimal. In working with patients/families to identify alternative care plans, as well as available community resources to assist with integrating a new diagnosis, the ED social worker serves as the link between the community, primary care, and hospital settings.

Language/Cultural Barriers

In one Michigan study, those diagnosed with cancer in the ED were found to come from lower socioeconomic backgrounds, were older, and more often disabled [13]. A large number of these patients were found to be dually eligible for Medicaid and Medicare up to 12 months prior to diagnosis compared to those diagnosed with cancer in other settings. Racial differences were also found in this study, with African Americans being significantly more likely to receive a cancer diagnosis in the ED [13].

A small study of Spanish-speaking patients in New York City found that they were frustrated both due to their inability to comprehend their prognosis, as well as difficulty in accurately expressing to clinicians their feelings surrounding cancer diagnosis and prognosis [15]. Advocating for the use of an interpreter for patients/families whose first language is not English can enhance understanding of medical information and minimize the emotional distress and confusion that emanates from such misunderstandings. An ED social worker's assessment of specific cultural concerns related to hospitalization, caregiving, and in some cases, receipt of medical treatment, allows the ED team to practice more culturally sensitive care.

A key tenet of social work practice is cultural competence or the ability to work in the context of cultural differences. Health disparities in ED pain management have been demonstrated, with one study showing that Hispanics were twice as likely not to receive pain medication for bone fractures as compared to non-Hispanic whites [16]. ED social workers can take the lead role in educating their interdisciplinary team members about these and other disparities as well as specific cultural concerns and traditions that may impact medical decision-making.

Caregiver Distress

Whether driven by distressing symptoms that are unmanageable at home, exhaustion from the intensity and/or longevity of caring for a cancer patient, or feeling overwhelmed by the responsibilities of caregiving, caregiver distress may be the root cause of an ED visit. ED social workers who screen for caregiver distress can work to determine additional sources of support in the caregiver's life and ensure a link to these services. A case example of an oncologic social work intervention for a distressed caregiver is provided at the end of this chapter.

The Role of the Outpatient Oncologic Social Worker

In addition to the acute medical needs that prompt ED visits and invite the interventions of ED social workers, oncologic social workers serve to address the unique psychosocial stressors that accompany a cancer diagnosis. These may include adjustment to a new cancer diagnosis, alterations in role and identity, changes in caregiver needs and family roles, impact on work and finances, and goals of care planning. For cancer patients, these transitions can be markers of ambiguous loss, or the unclear, indeterminate losses that are less acknowledged than death, but can greatly impact coping, sense of control, and psychosocial functioning in both the patient and family [17, 18]. These losses can also trigger feelings of anticipatory bereavement, in which patients and families begin the process of mourning, coping with loss, and psychosocial reorganization in preparation for death. This process can trigger mixed emotions such as helplessness, denial, confusion, and guilt but, if managed effectively, may provide patients and their families with improved communication and meaningful interactions at the end of life [19].

Prior to a cancer diagnosis, patients may not have had needs that would necessitate an interaction with a social worker. The oncologic social worker may be their first introduction to such services. Once this connection is made, the oncologic social worker can link patients and their families to concrete resources in the community, as well as provide therapeutic interventions to address psychosocial needs. The oncologic social worker becomes the conduit of communication, linking the work done in the ED with the work of the outpatient oncologic team; thus avoiding a "new beginning" for patients and families and supporting continuity of care in the process. It is the author's experience that oncologic and ED social workers typically collaborate and communicate to ensure smooth transitions for patients in need between specialties and across outpatient and inpatient settings.

Social workers are the primary psychosocial professionals available to patients receiving medical treatment [20, 21]. While medical social workers have become more broadly available in healthcare settings, oncologic social workers have evolved as a subspecialty within the field [22]. The psychosocial needs of patients with cancer have become increasingly complex as treatment has shifted to the outpatient setting. The broader range of treatment options available to patients has complicated decision-making and increased both patients' and families' care management responsibilities [23–25].

Traditional social work interventions in outpatient oncologic settings include biopsychosocial assessments, psychoeducation, counseling, linking patients to community resources and coordinating the provision of concrete services including home care, hospice, durable medical equipment, and transportation. Social workers are highly skilled practitioners who are trained to provide screening, assessment, and therapeutic interventions across the cancer continuum including primary prevention, diagnosis, treatment, survivorship, palliative care, end of life, and bereavement [22]. Oncologic social workers are knowledgeable about cancer and its treatments as well as psychosocial aspects of illness, cultural and spiritual influences, pain and symptom management, finances, community resources, and research in the field of psycho-oncologic [22, 26].

Clinical Interventions

Oncologic social workers spend considerable time with patients and/or family members discussing their adjustment to a cancer diagnosis. Individual counseling can help the patient determine specific concerns and set priorities [26, 27]. The focus of clinical work in health care is on enhancement of coping rather than psychopathology [28]. The goals of clinical interventions are to reduce anxiety and assist in clarifying misconceptions and correct misinformation, as well as decrease feelings of isolation [22, 29]. Researchers have shown that psychological interventions can improve the emotional and physical health outcomes in patients with cancer [29, 30].

Cognitive Behavioral Interventions

Cognitive behavioral therapy (CBT) combines cognitive psychotherapy with behavioral interventions. It seeks to reduce emotional distress by identifying, challenging, and eliminating irrational beliefs, and encouraging patients to change their maladaptive preconceptions and behaviors [31, 32]. These techniques may include hypnosis, guided imagery, progressive muscle relaxation, and biofeedback, which are utilized during individual or group sessions [26, 29]. Social workers frequently obtain specialized training for this type of work [26]. Used either alone or in conjunction with medication, behavioral methods are effective for treating side effects associated with cancer. These include anticipatory nausea and vomiting associated with treatment, heightened anxiety, and pain [33, 34].

Relaxation Techniques

Relaxation techniques guide patients to achieve control over their muscles and thoughts, in order to reduce emotional distress [31]. Progressive muscle relaxation involves systematic tensing and relaxing of various body parts. The practitioner describes comfortable sensations in muscle groups usually progressing from head down to feet or from feet up to head. Patients are encouraged to practice these techniques at home to enhance competence and achieve mastery [31]. Visualization of restful scenes associated with pleasurable thoughts is another technique used to promote a sense of relaxation and calmness, allowing patients to feel more in control of their feelings. Social workers practicing these techniques should obtain specific training in these areas [26]. The growing use of complementary therapies (such as meditation, relaxation, hypnosis, and visualization) has resulted in their increasing availability in hospitals and oncologic centers [35-37].

Supportive Counseling

Counseling helps patients and their families manage the multiple problems associated with chronic illness [26]. Individual supportive counseling can decrease the distress and disruption experienced with a cancer diagnosis. While there is no clear definition for supportive psychotherapy, this approach is generally considered an intervention that can be used intermittently or continuously. This patient-centered flexible approach assists patients in dealing with distressing emotions by reinforcing strengths [31, 38]. Supportive counseling emphasizes the importance of compassion, empathy, and support in working with patients [29]. An important goal of counseling in oncologic social work is to help the patient and/or family maintain or redefine hope [26], moving beyond equating hope with cure to broader meanings, values, and intentions that are beyond the limits of an illness.

Crisis Intervention

A significant crisis can be triggered with the initial cancer diagnosis, throughout the course of treatment, when the disease recurs, and at the termination of curative therapy. Therefore, oncologic social workers may use crisis intervention techniques on a recurring basis throughout the illness trajectory. Oncologic social workers help the patient and family explore and clarify feelings, understand how to manage these feelings, and teach new ways of coping, including solving problems [34, 39]. The oncologic social worker encourages familiar coping mechanisms, while also providing resources and support to patients experiencing a loss of stability [39].

Psychoeducation

Providing education to cancer patients serves to reduce the sense of helplessness that results from uncertainty and lack of knowledge. Psychoeducation can provide patients with a sense of mastery over their illness. It can involve disease-specific information and may also include information about coping, side effects, and wellness [29]. The goal of this intervention is to enhance coping skills and empower patients to become active participants in their care [29]. The information provided should be tailored to meet the patient's expectations, preferences, diagnosis, treatment, and prognosis. Information can also be related to maintenance of maximal health, coping, and financial/legal concerns.

A patient's educational needs change over time, such as during and after treatment [40]. A patient's primary language and reading comprehension level are also important factors for social workers to consider when adapting psychoeducational materials to cancer patients and their families. Oncologic social workers often provide patients with brochures, booklets, and materials from well-known sources, such as the Leukemia and Lymphoma Society, CancerCare, and the American Cancer Society, or refer patients to resource libraries and/or trusted websites. An understanding of patient health literacy is essential to effectively use these resources. Table 7.1 lists national resources that provide support, information, and/ or financial assistance to those with cancer and their families.

Table 7.1	Resources	for general	l cancer support
-----------	-----------	-------------	------------------

Resource	Website/phone
American Cancer Society	cancer.org 800-227-2345
American Psychosocial Oncologic Society	www.apos-society.org 866-276-7443
cancer.net	cancer.net 888-651-3038
CanCare	CanCare.org 888-461-0028
CancerCare	cancercare.org 800-813-HOPE
Cancer Hope Network	CancerHopeNetwork.org 800-552-4366
Cancer Financial Assistance Coalition	cancerFAC.org
Cancer Support Community/ Gilda's Club	CancerSupportCommunity.org 202-659-9709
Imerman Angels	Imermanangels.org 877-274-5529
Livestrong	Livestrong.org 855-220-7777
National Cancer Institute	Cancer.gov 800-4CANCER
National Comprehensive Cancer Network	nccn.org 215-690-0300

^aThis listing represents national organizations providing information to cancer patients. There are many more excellent disease-specific organizations and local organizations

Table 7.2 Description of psychosocial interventions that may be used with oncologic patients in the ED

	Intervention	Description	Outcome
1	Cognitive behavioral therapy	Assists patients in identifying and changing maladaptive thinking and behaviors, in order to reduce negative emotions and facilitate psychological wellness	Reduces anxiety, increases problem-solving skills, increases understanding of maladaptive cognitions, and enhances coping
	Relaxation techniques	Encompasses a variety of techniques to calm thoughts and muscles, in order to allow patients to feel more in control and at ease	Reduces anxiety, increases sense of control, enhances coping
	Supportive counseling	Focuses on helping patients to cope with distressing emotions, reinforces pre- existing strengths, and promotes adaptive adjustment to illness	Enhances coping by establishing a therapeutic alliance, reduces anxiety
	Crisis intervention	Time limited, used intermittently; focuses on symptom reduction; expression of feelings are encouraged and tangible support is provided	Reduces psychosocial symptoms, mobilizes social supports, increases sense of self-competency
	Psychoeducation	Utilizes educational resources and provides information to reduce feelings of helplessness while increasing the patient's knowledge and sense of control	Prevents ED admissions (i.e., may increase compliance to medical recommendations), fosters improved decision- making, reduces anxiety, increases sense of control

Cancer patients may benefit from numerous intervention techniques and programs. Psychiatric interventions play a significant role in the comprehensive care of cancer patients. The list provided in Table 7.2 is not an exhaustive list of interventions but rather a compilation of interventions that are utilized by oncologic social workers in practice, which may also be applicable to oncologic patients in the ED.

Case Discussion 2

Oncologic Social Worker Interventions to Prevent an ED Visit

Charlotte was a 57-year-old African American female with metastatic triple-negative breast cancer. Charlotte was well

known to her outpatient oncologic social worker (OSW), who had been working with her since she was initially diagnosed with cancer a year before. The OSW assisted with referral to a home hospice program which had an inpatient unit affiliated with the cancer center. Several days later, the OSW received a call from her adult daughter, "Julie." She was upset because the hospice nurse told her that Charlotte was entering the dying stage and they were unwilling to transfer her to the inpatient unit. Julie had a 4-year-old daughter at home and had been explicit at enrollment that she did not want her mother to die at home. She was very frustrated that the hospice staff was not helping her to facilitate an inpatient admission as they believed her symptoms could be managed at home. Julie stated that she was going to call an ambulance to bring Charlotte to a hospital, if the hospice agency did not transfer her mother immediately. In order to avoid the crisis of an ED visit and added distress for Charlotte and her daughter, the OSW intervened and advocated for an immediate inpatient admission since Julie was going to call 911/ ambulance if the agency did not act right away. The OSW also spoke to Julie and educated her about alternative options (other hospice programs with inpatient facilities), in order to avoid her calling 911/ambulance to have her mother brought to an ED. Within an hour, Charlotte was brought to the inpatient hospice unit. This case example demonstrates one way that continuity of relationship with an outpatient OSW, even in the setting of a hospice admission, assisted in preventing an ED visit/hospital admission and optimized the care of a dying patient, containing further risk of complicated bereavement for Julie and her 4-year-old child.

The Role of the Palliative Social Worker

Palliative social work developed as palliative care teams sought to increase patient and family-centered care for seriously ill patients and because of the need for the unique assessment and interventional skills provided by social workers. Building on the fields of hospice, oncologic, critical care, and other established areas of practice [41], early leaders in palliative social work helped identify specific competencies in palliative care, targeted psychosocial interventions, and areas of research [42]. Social workers are core interdisciplinary clinicians on palliative care teams whose multifaceted role includes education and counseling on one's adjustment to illness, with special attention to the multidimensional aspects of pain and other symptoms, including the impact of life-limiting illness on the patient's mood, goals, and relationships [43]. Palliative social workers also help facilitate patient/family decision-making regarding goals of care and advance care planning and provide therapeutic interventions to help reduce anxiety and distress in patients and families.

Palliative Social Work Initiative in the ED

As increasing numbers of ED's strive to integrate palliative care into their clinical setting with the goal of improving care for seriously ill patients and encouraging earlier palliative care consults and hospice referrals [44], there are opportunities for palliative social workers to provide consultation to patients, families, and clinicians in the ED. One study of ED utilization found that visits for palliative care, dehydration, and altered level of consciousness were higher during the final two-week period of life than during the last 6 months preceding death [45]. Palliative social workers may be called upon to consult in the ED to help patients and families integrate the meaning of the medical crises while attending to issues such as advance care planning, goals of care discussions and facilitating transitions in care.

The palliative social worker can obtain medical information, specify functional limitations in the patient, and complete symptom assessments [46], as well as discover the patient's narrative of the event that led them to the ED. The ability to ascertain unmet palliative care needs and communicate the benefits of a palliative consult or a hospice referral to the emergency physician encourages their initiation from the ED [47]. Early referral to palliative care in advanced cancer patients has shown to improve quality of life and does not appear to shorten survival [48]. The palliative social worker can be expected to communicate with the palliative care team regarding symptom management needs for seriously ill patients in the ED.

Visibility of the palliative care team is important in the eyes of emergency clinicians. The palliative social worker may become the "face" of palliative care by regularly consulting in the ED, through collaboration with the ED social worker to assist seriously ill patients, or by providing education on palliative care principles to ED clinicians. Whether helping to increase collaboration between emergency medicine and palliative care, identifying patients who are appropriate for palliative care consults and hospice referrals, or providing specific psychosocial interventions to patients in the ED [47], palliative social work can play a key role in assisting oncologic patients and families.

Goals of Care Conversations in the ED at End of Life

The ED is often where changes in the patient's illness trajectory are recognized, and new plans of care are established: thus identifying end-of-life patients who may be appropriate for a transition in care is appropriate in this setting. For oncologic patients who present to the ED and are actively dying, or for those whose prognosis is poor, facilitating goals of care discussions can help clarify options for ongoing diseasemodifying therapies. Within these discussions, it is important to understand the patient's or surrogate's wishes related to initiating, continuing, or foregoing potentially lifeprolonging treatment (e.g., endotracheal intubation). With the assistance of the emergency physician, the palliative social worker can help guide goals of care conversation to enhance patient and family's understanding of diagnosis, prognosis, and treatment options at the end of life, including palliative care and hospice.

The nature of an oncologic patient's emergency may also invite a conversation about advance directives or MOLST/ POLST to include decisions about resuscitation, treatment preferences, and goals of care [47]. Palliative social workers have the clinical skills and knowledge to work with the physician to obtain, interpret, and assist in completing these advance care-planning documents. ED social workers are also knowledgeable about advance care-planning documents as well as pertinent state laws that may pertain to this process.

Social Work Initiatives to Prevent ED Visits

Given the importance of ED social worker's role in care coordination, and other ED clinicians referral to them for assistance with illness adjustment [7], there is tremendous opportunity for them to be involved in improving care for cancer patients in the ED. Based on the acuity of cancer patients, initiatives will likely involve social workers who are most often responsible for making and following up on home care and hospice referrals. An initiative to assist in the prevention of ED visits requires screening, anticipatory guidance, and knowledge of resources. For example, caregivers experiencing distress and exhaustion might be connected to community resources that provide respite care. Below is a case example highlighting one such instance.

Case Discussion 3

OSW Connecting an Oncologic Patient's Caregiver to Community Resources

James is a 63-year-old, African American male with pancreatic cancer living in Tennessee. James's sister, Millie, his only caretaker, lives in New York City. When Millie found out about his diagnosis, she moved him to New York City, as he had no family in Tennessee. James sleeps on Millie's living room sofa and she assists him by coming to his medical appointments, chemotherapy appointments, making sure he eats properly, etc. Millie met with the oncologic social worker (OSW) to discuss her feelings of being overwhelmed by the level of care that he required and she indicated that she felt she needed help. The OSW offered to make a home

care referral and Millie agreed. After several weeks of home care, Millie reported that she was still feeling stressed and overwhelmed, and she requested that James be admitted to a nursing home. James has Medicaid, prompting the OSW to suggest that Millie try a day program at a nursing home close to where she lived. Millie agreed to the referral and James began attending the day program 4 days per week. The day program provided James with meals, activities, and transportation to and from Millie's apartment, all covered by his insurance. Millie was relieved by the referral, since the facility was one that she liked and was in her neighborhood. Millie had originally asked the oncologist for James to be admitted to the hospital in order to have him transferred to a skilled nursing facility. However, once the OSW suggested the day program, Millie no longer felt the need for James to stay full time at the nursing home. In this case, the OSW's involvement, recommendation, and referral to a community resource resulted in the avoidance of an ED admission.

Psychiatry

One initiative to avoid ED visits and an additional crisis for patients and families is through partnering or making psychiatry services readily available. The accessibility to a psychiatrist can be helpful in avoiding ED visits for suicide assessment(s). In outpatient cancer centers, if a psychiatrist is not available and a social worker determines that a patient is at risk, the patient would likely be sent to the ED for further assessment. This is not a beneficial allocation of ED resources as it is a potentially avoidable visit that can be upsetting for the patient. If a psychiatrist is on staff at the outpatient cancer center, or psychiatric services are readily available in the community, then ED visits to assess for suicidality may be avoidable.

Health Home Initiative

As more cancer patients are living longer with the disease, cancer is increasingly viewed as a chronic illness [49]. Due to the ongoing evolution of the American healthcare system and changes in reimbursement for services, hospitals will no longer be reimbursed at the same rates for readmissions and ED visits [50]. Therefore, many hospitals are making efforts to lower and contain costs through initiatives that have the potential to decrease readmissions and ED visits. The "Independence at Home" (IAH) initiative for Medicare beneficiaries who have been hospitalized and received rehabilitative services in the past 12 months target those with two or more chronic conditions. This health home initiative offers home-based primary care services aimed at reducing repetitive ED visits and hospitalizations. Under a collaborative

care model of multidisciplinary team members, one IAH initiative utilizes nurses and social workers to do the majority of home visits with the support of physicians [51]. Health homes are one example of a program in which social workers can play a pivotal role to reduce ED visits and readmissions for oncologic patients.

Suggestions for Future Research

More research is needed to understand the optimal role of ED social work in caring for oncologic patients, as well as to identify therapeutic interventions and their effectiveness with patients/families in enhancing their ED experience. There is also a need to identify collaborations and initiatives between ED, oncologic, and palliative social work that have the potential to strengthen the psychosocial care of the patient, avoid duplication of services, and possibly prevent unnecessary ED visits and hospitalizations. As research continues to be done to integrate palliative care in the setting of the ED, social workers, in the fields of emergency medicine, palliative care, and oncologic social work have an opportunity to provide expertise and leadership in the coordination of patient care.

Conclusion

Clinical social workers are trained to conduct a comprehensive biopsychosocial-spiritual assessment of patients and their families, to better inform goals of care, enhance communication, and ensure smooth transitions in care [6, 52]. In working with cancer patients who present to the ED in crisis, the ED social worker can identify psychosocial, financial, and cultural concerns that may impact future medical care. ED, oncologic, and palliative social workers are in key positions to help identify and test new initiatives aimed at strengthening services for oncologic patients throughout the continuum of their illness. Collaboration among these social workers allows for earlier outreach to oncologic patients with unmet psychosocial and concrete needs and supports continuity of care across settings. Attention to the psychosocial needs of oncologic patients and their families can help patients, caregivers, and medical providers optimize the delivery and efficacy of healthcare services while managing the emotional and social aspects of illness [40]. Bridging biomedical and psychological well-being aids in the promotion of better health.

The current healthcare climate focuses on market-driven, cost-containment strategies for the provision of medical care. As such, social workers serve an essential function in the cost-efficient delivery of health care. Social workers are well positioned to contribute to the psychosocial care of Acknowledgments The authors are very grateful to Terry Altilio, LCSW, and Halley Robinson, LMSW, for their constructive edits and assistance.

References

- Barbera L, Atzema C, Sutradhar R, Seow H, Howell D, Husain A, et al. Do patient-reported symptoms predict emergency department visits in cancer patients? A population-based analysis. Ann Emerg Med. 2013;61(4):427–37.
- Livingston PM, Craike M, Considine J. Unplanned presentations to the emergency departments due to chemotherapy induced complications: opportunities for improving service delivery. Australas Emerg Nurs J. 2011;14:62–8.
- Mayer DK, Travers D, Wyss A, Leak A, Waller A. Why do patients with cancer visit the emergency department? Results of a 2008 population study in North Carolina. J Clin Oncol. 2011;29(19):2683–8.
- Rondeau DF, Schmidt TA. Treating cancer patients who are near the end of life in the emergency department. Emerg Med Clin N Am. 2009;27(2):341–54.
- Van der Meer DM, Weiland TJ, Philip J, Jelinek GA, Boughey M, Knott J, et al. Presentation patterns and outcomes of patients with cancer accessing care in emergency departments in Victoria, Australia. Support Care Cancer. 2016;24:1251–60.
- Colon Y. End-of-life care. In: Gehlert S, Browne T, editors. Handbook of health social work. New York: Wiley; 2019. p. 572–85.
- Moore M, Whiteside LK, Dotolo D, Wang J, Ho L, Conley B, et al. The role of social work in providing mental health services and care coordination in an urban trauma center emergency department. Psychiatr Serv. 2016;67(12):1348–54.
- Lawson RR. Palliative social work in the emergency department. In: Altilio T, Otis-Green S, editors. Oxford textbook of palliative social work. London: Oxford University Press; 2011. p. 31–40.
- 9. Wells PJ. Preparing for sudden death: social work in the emergency room. Soc Work. 1993;38(3):3339–42.
- Selby S, Wang D, Murray E, Lang E. Emergency departments as the health safety nets of society: a descriptive and multicenter analysis of social worker support in the emergency room. Cureus. 2018;10(9):e3247.
- Takayesu JK, Hutson HR. Communicating life-threatening diagnoses to patients in the emergency department. Ann Emerg Med. 2004;43(6):749–55.
- 12. Bomba PA, Morrissey MB, Leven DC. Key role of social work in effective communication and conflict resolution process: medical orders for life-sustaining treatment (MOLST) program in New York and shared medical decision-making at the end of life. J Soc Work End Life Palliat Care. 2011;7:56–82.
- 13. Sikka V, Ornato JP. Cancer diagnosis and outcomes in Michigan EDs vs. other settings. Am J Emerg Med. 2012;30:283–92.
- 14. Henson LA, Higginson IJ, Daveson BA, Ellis-Smith C, Koffman J, Morgan M, et al. "I'll be in a safe place": a qualitative study of the decisions taken by people with advanced cancer to seek emergency department care. BMJ Open. 2016;6(11):e012134.
- Grudzen CR, Stone SS, Mohanty SA, Asch SM, Lorenz KA, Torres JM, et al. "I want to be taking my own last breath": patients' reflections on illness when presenting to the emergency department at the end of life. J Palliat Med. 2011;14(3):293–6.

- Todd KH, Samaroo N, Hoffman JR. Ethnicity as a risk factor for inadequate emergency department analgesia. JAMA. 1993;269:1537–9.
- 17. Betz G, Thorngren JM. Ambiguous loss and the family grieving process. Fam J. 2006;14:359–65.
- Boss P. Ambiguous loss: learning to live with unresolved grief. Cambridge: Harvard University Press; 1999. p. 1–25.
- Zilberfein F. Coping with death: anticipatory grief and bereavement. Generations. 1999;1:69–74.
- Coluzzi P, Grant M, Doroshow J, Rhiner M, Ferrell B, Rivera L. Survey of the provision of supportive care services at national cancer institute-designated cancer centers. J Clin Oncol. 1995;13(3):756–64.
- 21. Whitaker T, Weismiller T, Clark EJ. Assuring the sufficiency of a front line workforce: a national study of licensed social workers. Special report: social work services in health care settings. Washington, DC: National Association of Social Workers Center for Workforce Studies; 2006.
- 22. Kennedy V, Smolinski KM, Colon Y, Zabora J. Educating and training professional social workers in psycho-oncology. In: Holland JC, Breitbart WS, Butow PN, Jacobsen PB, Loscalzo MJ, McCorkle R, Butow P, editors. Psycho-oncology. 3rd ed. New York: Oxford University Press; 2015. p. 689–94.
- Fobair P, Stearns N, Christ G, Dozier-Hall D, Newman NW, Zabora J, et al. Historical threads in the development of oncology social work. J Psychosoc Oncol. 2009;27(2):155–215.
- 24. Hedlund S. Oncology social work: past, present and future. In: Christ G, Messner C, Behar L, editors. Handbook of oncology social work: psychosocial care for people with cancer. Oxford: Oxford University Press; 2015. p. 9–13.
- 25. Snow A, Warner J, Zilberfein F. The increase of treatment options at the end-of-life: impact on the social work role in an inpatient hospital setting. Soc Work Health Care. 2008;47(4):376–91.
- 26. Blum D, Clark E, Marcusen D. Oncology social work in the 21st century. In: Lauria MM, Clark EJ, Hermann JF, Stearns NM, editors. Social work in oncology: supporting survivors, families and caregivers. Atlanta: American Cancer Society; 2001. p. 45–71.
- Zebrak B, Jones B, Smolinski K. Oncology social work interventions throughout the continuum of cancer care. In: Chris G, Messner C, Behar L, editors. Handbook of oncology social work: psychosocial care for people with cancer. Oxford: Oxford University Press; 2015. p. 35–42.
- 28. Berkman B. The emerging healthcare world: implications for social work practice and education. Soc Work. 1996;41(5):541–51.
- Fawzy F, Fawzy N, Arndt L, Pasnau R. Critical review of psychosocial interventions in cancer care. Arch Gen Psychiatry. 1995;52:100–13.
- Moorey S, Watson M. Cognitive therapy. In: Holland JC, Breitbart WS, Butow PN, Jacobsen PB, Loscalzo M, McCorkle R, editors. Psycho-oncology. 3rd ed. New York: Oxford University Press; 2015. p. 458–63.
- Akechi T. Psychotherapy for depression among patients with advanced cancer. J Clin Oncol. 2012;42(12):1113–9.
- 32. Cagle J, Loscalzo M. Using cognitive behavioral approaches throughout the cancer experience. In: Christ G, Messner C, Behar L, editors. Handbook of oncology social work: psychosocial care for people with cancer. Oxford: Oxford University Press; 2015. p. 345–56.
- Neron S, Stephenson R. Effectiveness of hypnotherapy with cancer patients' trajectory: emesis, acute pain, and analgesia and anxiolysis in procedures. Int J Clin Exp Hypn. 2007;55:336–54.

- 34. Cox M, Stovall A. Social work interventions with children and adolescents. In: Lauria MM, Clark EJ, Hermann JF, Stearns NM, editors. Social work in oncology: supporting survivors, families and caregivers. Atlanta: American Cancer Society; 2001. p. 143–68.
- Paltiel O, Avitzour M, Peretz T, Cherny N, Kaduri L, Pfeffer RM, et al. Determinants of the use of complementary therapies by patients with cancer. J Clin Oncol. 2001;19(9):2439–48.
- Vickers A. Recent advances: complementary medicine. Br Med J. 2000;321(7262):683–6.
- Adams M, Jewell A. The use of complementary and alternative medicine by cancer patient. Int Semin Surg Oncol. 2007;4:10.
- Linn MW, Linn BS, Harris R. Effects of counseling for late stage cancers. Cancer. 1982;49:1048–55.
- Marcusen C. Transitions during cancer care. In: Christ G, Messner C, Behar L, editors. Handbook of oncology social work: psychosocial care for people with cancer. Oxford: Oxford University Press; 2015. p. 651–7.
- 40. Institute of Medicine (US) Committee on Psychosocial Services to Cancer Patients/Families in a Community Setting. In: Adler NE, AEK P, editors. Cancer care for the whole patient: meeting psychosocial health needs. Washington, DC: National Academies Press; 2008.
- Higgins PC. Guess who's coming to dinner? The emerging identity of palliative social workers. In: Altilio T, Otis-Green S, editors. Oxford textbook of palliative social work. London: Oxford University Press; 2011. p. 63–70.
- 42. Gwyther LP, Altilio T, Blacker S, Christ G, Csikai EL, Hooyman N, et al. Social work competencies in palliative and end-of-life care. J Soc Work End Life Palliat Care. 2005;1(1):87–120.
- 43. NASW. Standards of social work practice in palliative and end of life care [Internet]. National Association of Social Workers: Washington, DC; 2003. http://www.socialworkers.org/practice/ bereavement/standards/101503.asp?back=yes. Accessed 28 Apr 2014.
- 44. Meier DE, Beresford L. Fast response is key to partnering with the emergency department. J Palliat Med. 2007;10(3):641–5.
- 45. Barbera L, Taylor C, Dudgeon D. Why do patients with cancer visit the emergency department near the end of life? Can Med Assoc J. 2010;182(6):563–8.
- 46. Glajchen MG, Lawson RR, Homel P, DeSandre P, Todd KH. A rapid two-stage screening protocol for palliative care in the emergency department: a quality improvement initiative. J Pain Symptom Manag. 2011;42(5):657–62.
- Lawson RR. Palliative social work in the emergency department. J Soc Work End Life Palliat Care. 2012;8(2):120–34.
- 48. Grudzen CR, Richardson LD, Johnson PN, Hu M, Wang B, Ortiz JM, et al. Emergency department-initiated palliative care in advanced cancer. JAMA Oncol. 2016;2(5):591–8.
- Boeger-Knowles K, Ridley T. Chronic cancer: counseling the individual. Soc Work Health Care. 2014;53:11–30.
- Jacobs LR, Skocpol T. Health care reform and American politics: what everyone needs to know. New York: Oxford University Press; 2012. p. 17–49.
- Hostetter M, Klein S. In focus: making house calls to improve care of patients with advanced illness. 2014. http://www.commonwealthfund.org/Newsletters/Quality-Matters/2014/February-March/In-Focus.aspx.
- Altilio T, Otis-Green S. An emerging synergy: pain and social work. Newsline: quarterly insights edition. Natl Hosp Palliat Care Organ. 2007;18(6):31.

Part II

Prevention

© Springer Nature Switzerland AG 2021 K. H. Todd et al. (eds.), Oncologic Emergency Medicine, https://doi.org/10.1007/978-3-030-67123-5_8

8

Tobacco Control

Steven L. Bernstein

Background

Fifty years after Surgeon General Luther Terry's landmark report on smoking and lung cancer, tobacco use remains the leading cause of death in the United States and the leading cause of preventable death [1]. Worldwide, tobacco use is a growing cause of morbidity and mortality. In many developing countries, it is overtaking infectious diseases as a leading public health hazard. Although great progress has been made in curbing this man-made epidemic, the human and economic costs associated with smoking remain enormous.

In the United States, each year, about 437,000 Americans die from smoking [1]. An additional 41,000 die from exposure to secondhand smoke, largely as a result of living with a smoker. Although smoking prevalence has declined, in 2018, 13.8% of all Americans age 18 and older smoked [2]. The conditions associated with death from secondhand smoke exposure include lung cancer and coronary artery disease, residential fires, and prenatal and perinatal conditions such as sudden infant death syndrome [1]. Smoking is a causative agent in dozens of diseases, enumerated in Surgeon General reports dating back to 1964 and summarized in the most recent 2014 report [1]. These diseases are listed in Table 8.1. Of note, even half a century after publication of the first major Surgeon General's report on smoking, epidemiologic research continues to reveal new associations between smoking and certain cancers, such as renal cell carcinoma, pancreatic cancer, and acute myeloid leukemia.

Emergency departments, visited 138 million times by Americans in 2017 [3], are sites where ED clinicians may conduct opportunistic screening and intervention for tobacco use. ED-initiated tobacco control is effective as found by a 2017 meta-analysis [4] and can be done with a modicum of effort. The terms "smoking" and "tobacco use" are often used interchangeably. They are not. "Smoking" refers to the consumption of combustible tobacco. In the United States, that is largely in the form of cigarettes. Other forms of combustible tobacco include cigars, cigarillos, and hookah. In developing countries, bidi and kretek are also popular forms of consuming combustible tobacco.

Smokeless tobacco may be consumed as well, in the form of snus (moist pouched tobacco placed between the lip and gum), chewing tobacco, dip, and snuff (dried, insufflated tobacco). Newer products include nicotine-containing water.

Electronic cigarettes, which consist of a heating element that vaporizes a nicotine-containing solution, which is then inhaled, constitute a new and rapidly growing product. E-cigarettes, as they are known, come in a variety of delivery devices. Most solutions contain nicotine; some do not. There is no uniformity in the design or manufacture of these products, which now come under the regulatory purview of the US Food and Drug Administration (FDA) Center for Tobacco Products. The potential for e-cigarettes to cause illness, including cancer, cardiovascular disease, and addiction, is not well understood. They are currently the subjects of intense study, as well as substantial marketing efforts by the traditional tobacco companies, many of which have acquired e-cigarette manufacturers. Because of the developing science surrounding the health effects of e-cigarettes, they will not be discussed at length.

Diagnosis of Tobacco Use

Tobacco-related illness is common in the ED. A complete listing would include diseases directly caused by smoking, such as chronic obstructive pulmonary disease (COPD), and conditions like asthma whose acuity or treatment is complicated by co-occurring tobacco use. Table 8.1 summarizes the list of tobacco-caused illnesses. An early paper found that about 5% of all ED visits, 7% of all admissions, and 10% of ED charges are attributable to smoking [5].

S. L. Bernstein (🖂)

Department of Emergency Medicine, Yale New Haven Hospital, Yale University School of Medicine, New Haven, CT, USA e-mail: steven.bernstein@yale.edu

Table 8.1	Relative risks for adult mortalit	y from smoking-related o	diseases, adults 35 yea	ars of age and older, bas	sed on Cancer Prevention	on Study
II, United	States					

	Males		Females	
Disease category (ICD-10 code)	Current smoker	Former smoker	Current smoker	Former smoker
Malignant neoplasms				
Lip, oral cavity, pharynx (C00–C14)	10.89	3.40	5.08	2.29
Esophagus (C15)	6.76	4.46	7.75	2.79
Stomach (C16)	1.96	1.47	1.36	1.32
Pancreas (C25)	2.31	1.15	2.25	1.55
Larynx (C32)	14.60	6.34	13.02	5.16
Trachea, lung, bronchus (C33–C34)	23.26	8.70	12.69	4.53
Cervix uteri (C53)	n/a	n/a	1.59	1.14
Kidney and renal pelvis (C64-C65)	2.72	1.73	1.29	1.05
Urinary bladder (C67)	3.27	2.09	2.22	1.89
Acute myeloid leukemia (C92.0)	1.86	1.33	1.13	1.38
Cardiovascular diseases				
Coronary heart disease (I20-I25)				
Persons 35-64 years of age	2.80	1.64	3.08	1.32
Persons ≥65 years of age	1.51	1.21	1.60	1.20
Other heart disease (I00–I09, I26–I28, I29–I51)	1.78	1.22	1.49	1.14
Cerebrovascular disease (I60–I69)				
Persons 35-64 years of age	3.27	1.04	4.00	1.30
Persons ≥ 65 years of age	1.63	1.04	1.49	1.03
Atherosclerosis (I70)	2.44	1.33	1.83	1.00
Aortic aneurysm (I71)	6.21	3.07	7.07	2.07
Other arterial disease (I72–I78)	2.07	1.01	2.17	1.12
Respiratory diseases				
Influenza, pneumonia (J10–J11, J12–J18)	1.75	1.36	2.17	1.10
Bronchitis, emphysema (J40–J42, J43)	17.10	15.64	12.04	11.77
Chronic airways obstruction (J44)	10.58	6.80	13.08	6.78

From 2014 Surgeon General's report [1]

Note: ICD = International Classification of Diseases

Emergency physicians and nurses screen for smoking irregularly. Tobacco use is more likely to be solicited for patients with conditions that are clearly tobacco-related; less so for others.

There are various ways to screen for tobacco use. In research contexts, a two-question screener is often used. The screener is used by two large annual surveys, managed by the Centers for Disease Control and Prevention: the Behavioral Risk Factor Surveillance System (BRFSS) and the National Health Interview Survey (NHIS).

The two questions are:

- 1. Have you smoked at least than 100 cigarettes in your entire life?
 - 🗌 No
 - ☐ Yes
 - Don't know/not sure
 - □ Refused
- 2. Do you now smoke cigarettes every day, some days, or not at all?
 - Every day
 - □ Some days

- 🗌 Not at all
- Don't know/not sure
- □ Refused

Individuals who endorse having smoked at least 100 cigarettes/lifetime, and are every- or some-day smokers, are considered to be current smokers. Individuals who endorse at least 100 cigarettes/lifetime, but do not currently smoke, are considered to be former smokers. Those smoking less than 100 cigarettes/lifetime are considered never-smokers.

Of note, these questions do not capture the use of other forms of combustible tobacco, or smokeless forms such as chew, snus, and the newer heat-not-burn cigarettes. Electronic cigarettes (e-cigarettes) and related products, known collectively as electronic nicotine delivery systems (ENDS), constitute a new and growing means of nicotine administration. The oncogenic and pathogenic potential of ENDS is only starting to be studied, although the market share of these products is growing rapidly.

However, in the context of routine clinical care, it is probably sufficient to ask patients if they currently smoke. In our experience, smokers tend to be forthcoming in disclosing their tobacco use. In the current era of data capture via electronic health records (EHRs), there is typically a structured field in the social history (or elsewhere) to record smoking status. In that case, the provider's choices may be constrained by the responses offered in the "smoking box" of the EHR.

Diagnosis of Tobacco-Related Illness

The list of conditions in Table 8.1 is extensive but does not cover all clinical scenarios in which EM practitioners might discuss smoking with patients. For example, wound healing is often compromised in smokers, with higher risks of poor cosmesis and infection [6]. Injury comprises 18.9% of all ED visits [3], so smokers with injuries are common. Tobacco abstinence should be advised for all smokers with lacerations, fractures, abscesses, and other skin, soft tissue, and musculoskeletal injuries. Discharge summaries generated by electronic medical records should mention tobacco avoidance for patients with traumatic injury.

Illnesses Associated with Tobacco Use

The number of diseases associated with tobacco use is substantial, and Surgeon General reports since 1964 continue to identify new conditions associated with smoking. The list of tobacco-related illnesses, along with their associated relative risks for mortality, is summarized in Table 8.1.

Note that many of these conditions are commonly seen in the ED. These are largely the cardiovascular diseases, such as chest pain, acute coronary syndromes including myocardial infarction and unstable angina pectoris, and respiratory diseases including pneumonia, influenza, exacerbations of chronic bronchitis and emphysema, and asthma. Patients with cancer are, of course, seen in the ED. They generally present with a complication of treatment or the cancer itself.

Cancer is occasionally, albeit rarely, diagnosed de novo in the ED. It is important to note that these diagnoses are presumptive, because no tissue diagnosis has yet been made.

Some possible scenarios in which cancer may be presumptively diagnosed include:

- A heavy smoker who presents with a cough, dyspnea, or weight loss and has a new pulmonary mass seen on chest x-ray
- A heavy smoker who presents with marked weight loss, progressive difficulty swallowing, and a mediastinal mass contiguous with the esophagus seen on chest x-ray or CT scan
- A postmenopausal woman who presents with vaginal bleeding

- 123
- A person who presents with fever and generalized bleeding and is found to be thrombocytopenic with many blast cells in the peripheral blood smear

Of note, tobacco use also is relevant in the ED management of conditions not formally associated with smoking. For example, acute exacerbations of asthma are commonly treated in the ED [7]. Although asthma is not caused by smoking, tobacco use is common among ED asthmatics. It increases the frequency and severity of attacks and prolongs the duration of the exacerbation. Another example would be the management of cellulitis in the foot of a person with diabetes and concurrent smoking. Smoking contributes to the development of peripheral vascular disease and may impair wound healing.

Emergency Department Treatment of Tobacco Dependence

Because of tobacco's great burden of illness and death, its disproportionate use by individuals of low socioeconomic status (SES), and the heavy use of EDs by low SES individuals, the ED has been regarded as an opportune venue in which to initiate treatment for smoking. The accumulating evidence of tobacco's causative role in ED-managed illness and injury has led to its inclusion in the core curriculum of the specialty [8]. Much of the research in this area has entailed understanding provider facilitators and barriers to ED-initiated interventions for smoking.

The general approach to ED-initiated intervention for smoking is adapted from the model known as Screening, Brief Intervention, and Referral to Treatment (SBIRT) [9]. SBIRT entails using one or two questions to identify an individual with a risky health behavior, offering an abbreviated form of motivational interview [10] to promote behavior change and then referring to an appropriate source of aftercare. Initially developed to identify and intervene with persons with alcohol use disorders, SBIRT has been endorsed by the Substance Abuse and Mental Health Services Administration and other professional bodies for use in the ED [11]. ED-based studies with more intensive interventions have generally offered a combination of SBIRT (tailored for smokers) and motivational interviewing.

A newer approach to ED-initiated interventions known as Screening, Treatment Initiation, and Referral (STIR) [12] incorporates medication management into the ED intervention. In the case of smoking, a STIR-informed approach would include the initiation in the ED of nicotine replacement therapies such as patches and gum. Several studies have demonstrated the efficacy of STIR for tobacco dependence [13, 14] and buprenorphine for managing opioid dependence [15]. STIR capitalizes on the teachable moment often found in the ED visit, when patients may be motivated to change an unhealthy behavior [16].

There are numerous evidence-based treatments for tobacco dependence. These may be divided into two broad categories: medication and counseling. Each is effective; used in combination, they provide even greater efficacy.

There are seven FDA-approved medications: nicotine patch, gum, lozenge, nasal spray, and inhaler, plus varenicline and bupropion. Counseling strategies with proven efficacy include one-on-one in-person sessions, group counseling, and telephone quitlines. The evidence base supporting these treatments is reviewed extensively in the 2008 Public Health Service's guideline on tobacco dependence treatment and in the 2014 Surgeon General's report on smoking.

Of note, quitlines are widely available in all 50 states. They can be accessed by a single phone number: 1-800-QUIT-NOW. Services vary somewhat from state to state, but as a rule include counseling by a trained provider, provision of written materials, starter doses of nicotine replacement, webbased services, and, increasingly, smartphone-based texting services. Quitlines are open 7 days a week, and languages other than English are available. Referrals can be made by providers or smokers. There is no cost to individuals or health systems, and insurance is not needed. Additional information is available at www.naquitline.org, the home page of the North American Quitline Consortium.

Most smoking cessation counseling uses principles of motivational interviewing or cognitive behavioral therapy. Of note, neither hypnosis nor acupuncture has demonstrated efficacy.

These treatments are summarized in Table 8.2.

The pharmacotherapy of nicotine dependence treatment is straightforward. Smokers who consume five or more cigarettes daily are good candidates for treatment. Medication is typically begun with a single agent, usually the nicotine patch or gum. A single cigarette contains 1–3 mg of nicotine, which can be used to guide dosing. In general, nicotine should be replaced milligram-for-milligram. A 21-mg patch, applied daily, would be a typical treatment for someone who smokes ten or more cigarettes daily. Higher dosing or additional forms of nicotine replacement therapy (NRT) may be added if the patient experiences cravings. Recent studies suggest combination therapy, using both long-acting and short-acting agents (e.g., patch and gum or nasal spray or inhaler), may be more effective than monotherapy. The reason is that transdermal nicotine generally does not replace enough nicotine to prevent cravings and other symptoms of withdrawal. NRT products that cross the blood-brain barrier quickly and easily can offer rapid relief for smokers with cravings.

Bupropion is a drug whose mechanism of action is incompletely understood. It was initially approved for treatment of mood disorders but also shows efficacy in smoking cessation. Varenicline is an interesting drug that blocks nicotinic receptors in the brain that mediate reward and craving. It is an agonist-antagonist. Varenicline prevents nicotine from binding to receptors but stimulates the release of a small amount of dopamine, generally sufficient to prevent symptoms of withdrawal. These drugs are beyond the scope of practitioners of emergency care and are not indicated for initiation in the ED.

The clinical trials of ED-initiated tobacco dependence treatment are summarized in Table 8.3 [13, 17–23, 25]. Most were single-institution studies with modest sample sizes and limited methodologic rigor. More recent studies from the 2010s used larger sample sizes, biochemical confirmation of tobacco abstinence at follow-up, and rigorous methods to minimize attrition and found various interventions to be efficacious.

A study from 2015 [13] found that a multicomponent intervention was able to produce a statistically significant higher rate of tobacco abstinence in subjects at the primary endpoint, 3 months, compared to controls. At 1 year, the effect attenuated but nearly reached statistical significance. The intervention consisted of provision of 6 weeks of nicotine patches and gum, initiation of the patch in the ED, a brief motivational interview (10-15 min) by a trained interventionist, a referral faxed to the state smokers' quitline, a phone call 2-3 days after enrollment, and a smoking cessation brochure. This study was the first to demonstrate the efficacy of ED-initiated tobacco dependence treatment. Although efficacious, the intervention has limited generalizability because of the use of nonclinical personnel to perform the motivational interview and the provision of a substantial supply of nicotine replacement medication.

An additional limitation of this study was the inability to disaggregate the effects of individual components, given the intervention was delivered as a package. To identify the efficacy of individual components, as well as interactions between components, the investigators conducted а $2 \times 2 \times 2 \times 2$ factorial trial using the multiphase optimization strategy (MOST) [27]. MOST studies use iterative approaches to identify packages of effective intervention components, subject to a cost constraint. In this study, 1056 adult smokers were enrolled and randomized to receive up to four interventions: a brief motivational interview, 6 weeks of nicotine gum and patches with the first dose applied in the ED, an active referral to a smokers' quitline, and enrollment in the SmokefreeTXT texting program of the National Cancer Institute. Each intervention was "on" or "off" for each participant, so the trial had 16 conditions.

At 3 months' follow-up, both the motivational interview and the nicotine replacement therapy were found to increase abstinence (motivational interview 13.5% vs. 8.9% [P = 0.02]; NRT, 14.4% vs. 8.0% [P = 0.001]) [14]. Neither

Table 8.2 Tobacco depender	ice treatment medications
----------------------------	---------------------------

Products OTC	Dosage	Duration	Precautions	Adverse effects	Patient education
Nicotine patch 21 mg 14 mg 7 mg	One patch per day. >10 cpd: 21 mg 4 weeks, 14 mg 2 weeks ≤10 cpd: 14 mg 4 weeks 7 mg 2 weeks	8–12 weeks	Do not use if patient has severe eczema or psoriasis Caution within 2 weeks of MI	Local skin reaction Insomnia	Apply each day to clean, dry, hairless skin Focal rash is common: Rotate site daily Available without prescription.
Nicotine gum 2 mg 4 mg	First cigarette ≤ 30 min after waking: 4 mg First cigarette > 30 min after waking: 2 mg 1 piece every 1–2 h	12 weeks	Caution with dentures Do not eat or drink 15 min before or during use Limit 24 in 24 h	Mouth soreness Stomachache Hiccups	DO NOT CHEW LIKE ORDINARY GUM Alternate chewing and "parking" between the cheek and gum (chew until mouth tingles and then park for 1 min, continue for 30 min) Nicotine absorbed across the buccal mucosa Avoid food and acidic drinks before and during use Available without prescription
Nicotine lozenge 2 mg 4 mg	First cigarette ≤ 30 min after waking: 4 mg First cigarette > 30 min after waking: 2 mg 1 piece every 1–2 h	12 weeks	Do not eat or drink 15 min before use One lozenge at a time Limit to 20 in 24 h	Heartburn Local irritation of the mouth and throat Coughing Hiccups	DO NOT BITE, CHEW, or SWALLOW Dissolve in mouth slowly Each lozenge takes 20–30 min to dissolve Avoid food and acidic drinks before and during use Available without prescription
Nicotine inhaler Nicotrol Inhaler®	6–16 cartridges/ day Each cartridge = 2 cigs Use 1 cartridge q 1–2 h	6 months; taper	Reactive airway disease	Mouth and throat irritation Cough	Patient is not to puff like a cigarette. Gentle puffing recommended Absorption via the buccal mucosa Avoid food and acidic drinks before and during use
Nicotine nasal spray Nicotrol NS®	1–2 sprays each nostril/h 8–40 doses/day	3–6 months; taper	Not for patients with asthma	Nasal irritation Sneezing Cough Teary eyes	Instruct patient to tilt the head back and spray Tolerance to local adverse effects develops first week after use
Bupropion SR150 Zyban® or Wellbutrin®	Start 1–2 weeks before quit date Days 1–3: 150 mg each morning Days 4–end: 150 mg BID	2–6 months	Contraindications: Seizure disorder Current use of MAO inhibitor Eating disorder Alcohol dependence Head trauma	Insomnia Dry mouth Anxiety	Take second pill early evening to reduce insomnia Never double dose if miss a pill
Varenicline Chantix®	Start 1 week before quit date 0.5 mg/day for 3 days and then 0.5 mg BID for the next 4 days After first 7 days 1 mg/BID	3–6 months	Persons with kidney problems require dose adjustment Serious psychiatric illness	Nausea Insomnia Abnormal dreams	Take after eating and with water (full glass) Never double dose. Take missed dose as soon as remembered. If close to next dose wait and take at regular dose time Nausea is usually transient. If nausea persists, dose reduction is recommended

OTC over the counter, MI myocardial infarction, MAO monoamine oxidase

the quitline nor the texting program was efficacious. No interactions were noted. This study was the first to identify individually effective components of ED-initiated tobacco control. Future work for ED-based tobacco treatment should focus on the scalability of these interventions. Table 8.4 reviews the components of an effective ED-initiated tobacco intervention. The individual components are all supported by evidence from high-grade clinical trials in various settings, with at least one high-quality ED trial to support their use.

	Mantel-Haenszel relative risk (95% confidence interval)				
Year of publication, study	1 month	3 months	6 months	12 months	
2000, Antonacci and Eyck	-	-	0.33 (0.01-7.74) ^b	-	
[17]					
2000, Richman et al. [18]	-	1.14 (0.36–3.57)	-	-	
2007, Horn et al. [19]	-	-	0.83 (0.05–12.77)	-	
2007, Schiebel and Ebbert	-	2.00 (0.20-20.33)	9.00 (0.52-156.91) ^b	-	
[20]					
2008, Bock et al. [21]	1.64 (1.04–2.56)	1.35 (0.86–2.12)	1.04 (0.64–1.68)	-	
2008, Boudreaux et al. [22]	-	1.86 (0.25–13.91)	-	-	
2009, Neuner et al. [23]	1.30 (0.79–2.15)	1.13 (0.75–1.69)	1.14 (0.81–1.61)	1.25 (0.91–1.72)	
2011, Anders et al. [24]	-	0.62 (0.21–1.83)	-	-	
2011, Bernstein et al. [25]	-	1.12 (0.66–1.91)	-	-	
2013, Cheung et al. [26]	1.69 (0.56-5.08)	1.93 (0.66–5.63)	0.64 (0.27–1.55)	0.55 (0.18-1.66)	
2015, Bernstein et al. [13]	-	2.49 (1.49-4.16)	-	1.38 (0.97–1.98)	
Meta-analyses	1.49 (1.08-2.05)	1.38 (1.12–1.71)	1.09 (0.84–1.41)	1.26 (1.00–1.59)	
	[P = 0.01]	[P = 0.003]	[P = 0.54]	[P = 0.05]	

Table 8.3 Benefit of emergency department-initiated tobacco control compared with control condition on tobacco-use results of individual studies (n = 11) and meta-analyses, by follow-up time^a

Adapted from Lemhoefer et al. [4]

^aThis systematic review and meta-analysis updates a previous review and includes publications published between October 4, 2010, and May 15, 2015

 $^{b}0.5$ added to all cells of the 2 × 2 table in calculating the relative risks to avoid degeneracy caused by sampling zero counts

Table 8.4	Components	of an	effective	e ED-initiated	intervention	for tobacco	dependence
-----------	------------	-------	-----------	----------------	--------------	-------------	------------

Component	Comments
Counseling	Brief counseling intervention employing principles of motivational interviewing; cognitive behavioral treatment may be efficacious
Medication	Provision of at least 4 weeks of nicotine replacement therapy. Combining short- and long-acting forms (e.g., patch and gum) likely to be more efficacious than monotherapy
Post-discharge treatment:	Aftercare should extend at least 30 days beyond visit. Active referral to state smokers' quitline, via fax or electronic
quitline, texting	health record, may achieve that. Newer interventions such as use of cellphone texting warrant further study
Interventionist	Ideally, a nonclinical individual, such as a health promotion advocate or health educator. Can be delivered by physicians, midlevel providers, and nurses, but constraints of time and clinical burden are substantial

Cost

Tobacco dependence treatment is among the most inexpensive, most cost-effective interventions in clinical medicine [28]. Integrating tobacco dependence screening, treatment, and referral into ED clinical workflows can be quite inexpensive. Several models of practice are available. The cheapest is to allow providers-physicians, nurses, and midlevel practitioners-to perform the screening as part of routine clinical care. Brochures advertising the state tobacco quitline, generally available from health departments in bulk at little to no cost, can be distributed to smokers. Advice to quit, call the quitline, or perhaps visit a locally available smoking cessation clinic can be templated and added to discharge summaries. Directed referrals to quitlines via fax can be made by clinical or clerical personnel. Some electronic medical records are integrating quitline referrals into their order sets for tobacco dependence [29].

A more intensive, and expensive, model of care entails placing lay educators or health promotion advocates in EDs to screen patients for tobacco use and other risky health behaviors [30]. These models are effective in identifying and referring patients, but their impact on long-term abstinence rates is unclear.

Conclusion

Tobacco use is widely prevalent in emergency department patients, and tobacco-related illness is a common reason for presentation. Recent evidence suggests that ED initiation of nicotine replacement therapy and behavioral counseling are independently effective in promoting sustained tobacco abstinence. As a result of the accumulating evidence regarding the efficacy of ED-initiated tobacco control, both the US Public Health Services' clinical practice guideline [31] and a report by the Institute of Medicine [32] recommend EDs as effective loci for tobacco screening and treatment. Tobacco use carries a sufficient burden of illness and death to warrant routine screening and intervention in ED patients. Future work should focus on finding ways to disseminate and implement these effective interventions.
References

- U.S. Department of Health and Human Services. The health consequences of smoking—50 years of progress. A report of the surgeon general. Atlanta: U.S. Department of Health and Human Services CfDCaP, National Center for Chronic Disease Prevention and Health Promotion, Office on Smoking and Health; 2014.
- Centers for Disease Control and Prevention, National Center for Health Statistics. Summary Health Statistics: National Health Interview Survey, 2018. Table A-12a. Age-adjusted percentages (with standard errors) of current cigarette smoking status among adults aged 18 and over, by selected characteristics: United States, 2018. 2019. https://ftp.cdc.gov/pub/Health_Statistics/ NCHS/NHIS/SHS/2018_SHS_Table_A-12.pdf. Accessed 1 Nov 2019.
- National Center for Health Statistics, Centers for Disease Control and Prevention. National Hospital Ambulatory Medical Care Survey: 2017 Emergency Department Summary Tables. 2017. https://www.cdc.gov/nchs/data/nhamcs/web_tables/2017_ed_ web_tables-508.pdf. Accessed 13 Jun 2020.
- Lemhoefer C, Rabe GL, Wellmann J, Bernstein SL, Cheung KW, McCarthy WJ, et al. Emergency department-initiated tobacco control: update of a systematic review and meta-analysis of randomized controlled trials. Prev Chronic Dis. 2017;14:E89.
- Bernstein SL. The impact of smoking-related illness in the ED: an attributable risk model. Am J Emerg Med. 2002;20(3):161–4.
- Silverstein P. Smoking and wound healing. Am J Med. 1992;93(1 Suppl 1):S22–4.
- Silverman RA, Boudreaux ED, Woodruff PG, Clark S, Camargo CA Jr. Cigarette smoking among asthmatic adults presenting to 64 emergency departments. Chest. 2003;123(5):1472–9.
- Council of Residency Directors in Emergency Medicine. 2019 Model of the clinical practice of emergency medicine. https://www. cordem.org/globalassets/files/misc.-files/2019-em-model_website. pdf. Accessed 13 Jun 2020.
- Babor TF, McRee BG, Kassebaum PA, Grimaldi PL, Ahmed K, Bray J. Screening, Brief Intervention, and Referral to Treatment (SBIRT): toward a public health approach to the management of substance abuse. Subst Abus. 2007;28(3):7–30.
- Miller WR, Rollnick S. Motivational interviewing: preparing people for change. 2nd ed. New York: Guilford Press; 2002.
- Substance Abuse and Mental Health Services Administration. Screening, brief intervention, referral, and treatment. https://www. samhsa.gov/sbirt. Last updated 15 Sept 2017. Accessed 10 Aug 2020.
- Bernstein SL, D'Onofrio G. Screening, treatment initiation, and referral for substance use disorders. Addict Sci Clin Pract. 2017;12(1):18.
- Bernstein SL, D'Onofrio G, Rosner J, O'Malley S, Makuch R, Busch S, et al. Successful tobacco dependence treatment in lowincome emergency department patients: a randomized trial. Ann Emerg Med. 2015;66(2):140–7.
- 14. Bernstein SL, Dziura J, Weiss J, Harper Brooks A, Miller T, Vickerman KA, et al. Successful optimization of tobacco dependence treatment in the emergency department: A randomized trial using the Multiphase Optimization Strategy. Poster presentation at the Annual Meeting of the Society for Research on Nicotine and Tobacco, New Orleans, LA; March 13, 2020.
- D'Onofrio G, O'Connor PG, Pantalon MV, Chawarski MC, Busch SH, Owens PH, et al. Emergency department-initiated buprenor-

phine/naloxone treatment for opioid dependence: a randomized clinical trial. JAMA. 2015;313(16):1636–44.

- Boudreaux ED, Baumann BM, Camargo CA Jr, O'Hea E, Ziedonis DM. Changes in smoking associated with an acute health event: theoretical and practical implications. Ann Behav Med. 2007;33(2):189–99.
- Antonacci MA, Eyck RT. Utilization and effectiveness of an emergency department initiated smoking cessation program (abstract). Acad Emerg Med. 2000;7(10):1166.
- Richman PB, Dinowitz S, Nashed AH, Eskin B, Sylvan E, Allegra C, et al. The emergency department as a potential site for smoking cessation intervention: a randomized, controlled trial. Acad Emerg Med. 2000;7(4):348–53.
- Horn K, Dino G, Hamilton C, Noerachmanto N. Efficacy of an emergency department-based motivational teenage smoking intervention. Prev Chronic Dis. 2007;4(1):A08.
- Schiebel N, Ebbert J. Quitline referral vs. self-help manual for tobacco use cessation in the emergency department: a feasibility study. BMC Emerg Med. 2007;7(1):15.
- Bock BC, Becker BM, Niaura RS, Partridge R, Fava JL, Trask P. Smoking cessation among patients in an emergency chest pain observation unit: outcomes of the Chest Pain Smoking Study (CPSS). Nicotine Tob Res. 2008;10(10):1523–31.
- Boudreaux ED, Baumann BM, Perry J, Marks D, Francies S, Camargo CA Jr, Ziedonis D. Emergency department initiated treatments for tobacco (EDITT): a pilot study. Ann Behav Med. 2008;36(3):314–25.
- Neuner B, Weiss-Gerlach E, Miller P, Martus P, Hesse D, Spies C. Emergency department-initiated tobacco control: a randomised controlled trial in an inner city university hospital. Tob Control. 2009;18(4):283–93.
- Anders ME, Sheffer CE, Barone CP, Holmes TM, Simpson DD, Duncan AM. Emergency department-initiated tobacco dependence treatment. Am J Health Behav. 2011;35(5):546–56.
- Bernstein SL, Bijur P, Cooperman N, Jearld S, Arnsten JH, Moadel A, Gallagher EJ. A randomized trial of a multicomponent cessation strategy for emergency department smokers. Acad Emerg Med. 2011;18(6):575–83.
- 26. Cheung K, Wong I, Brubacher J, Abu-Laban R, Schulzer M. Emergency department tobacco cessation counseling: implementation and evaluation of a community-based program: a pilot study (abstract). Can J Emerg Med. 2013;15(S1):S3.
- Collins LM. Optimization of behavioral, biobehavioral, and biomedical interventions: the multiphase optimization strategy (MOST). Cham: Springer International; 2018.
- Parrott S, Godfrey C. Economics of smoking cessation. BMJ. 2004;328(7445):947–9.
- 29. Adsit RT, Fox BM, Tsiolis T, Ogland C, Simerson M, Vind LM, et al. Using the electronic health record to connect primary care patients to evidence-based telephonic tobacco quitline services: a closedloop demonstration project. Transl Behav Med. 2014;4(3):324–32.
- D'Onofrio G, Degutis LC. Integrating project ASSERT: a screening, intervention, and referral to treatment program for unhealthy alcohol and drug use into an urban emergency department. Acad Emerg Med. 2010;17(8):903–11.
- Agency for Healthcare Research and Quality. Treating tobacco use and dependence: 2008 update. Content last reviewed February 2020. Rockville: Agency for Healthcare Research and Quality. https://www.ahrq.gov/prevention/guidelines/tobacco/index.html.
- Bonnie RJ, Stratton K, Wallace R. Ending the tobacco problem: a blueprint for the nation. Washington, DC: National Academies Press; 2007.

Problem Drinking

Edouard Coupet Jr. and Federico E. Vaca

Case Study

Linda is a 52-year-old woman who presents to the emergency department (ED) with a chief complaint of a forehead laceration following a fall down a flight of stairs. While she was at home, she states that she had consumed a half a pint of vodka in the afternoon. While traveling down the stairs, she missed the last step, fell, struck her head, and sustained a small laceration to her forehead. She denies any other complaints and otherwise is nontoxic appearing. Her past medical history includes breast cancer, which was diagnosed 4 years ago, and hypertension. While in the ED, the physician orders a CT scan of her head which found no evidence of acute intracranial injury. The nurse administers tetanus prophylaxis and the physician repairs Linda's laceration. While repairing her laceration, the physician finds out that Linda drinks at least a pint of vodka every day. The physician soon determines that she appears clinically sober and almost ready for discharge. However, there is concern about Linda's heavy alcohol use. The physician notifies the social worker who comes to talk to Linda about her alcohol use and drinking pattern. A medical chart review reveals that Linda has presented to the ED three times in the last year for various falls. After further brief discussion and evaluation, it is determined that Linda meets the criteria for moderate alcohol use disorder (AUD). The social worker informs and educates Linda about unhealthy drinking and drinking patterns that are associated with AUDs. She further assesses her willingness to seek AUD treatment. Linda states she is not interested in seeking inpatient treatment at this time but is willing

Funding: K12DA033312-08 from the National Institute on Drug Abuse (Coupet & Vaca).

E. Coupet Jr. $(\boxtimes) \cdot F$. E. Vaca

to accept a referral for outpatient treatment services. Linda is discharged from the ED and provided a follow-up appointment at a specialized treatment clinic that can help manage her AUD.

Background

Alcohol consumption plays a substantial role in human culture worldwide [1, 2]. It has been estimated to account for approximately 4% of the total global disease burden [3]. In 2016, it was the seventh leading risk factor for death, globally [4]. In the USA, similar to many other industrialized countries, alcohol use disorders (AUDs) remain among the most common, yet undertreated, behavioral health disorders [5]. This is particularly important because individuals with AUD contribute to nearly half of alcohol-related diseases, including various cancers and injuries [6].

Alcohol use most commonly begins during adolescence [7–10]. As youth progress from early to late adolescence, alcohol use typically increases. According to data from the Monitoring the Future National Survey Results on Drug Use, in 2018, 8% of 8th graders reported alcohol use, while 30% of 12th graders reported alcohol use in the past month [11]. Moreover, the earlier youth begin to use alcohol, the higher their risk is for alcohol-related adverse consequences. In one study of over 27,000 current and former individuals who have drank, an estimated 40% of individuals who started drinking at age 14 or younger developed alcohol dependence over their lifetime. The rate declined to nearly 10% when the individual started drinking at age 20 or older [12].

Previous literature has characterized four individual trajectory groups (i.e., low, high, increasing, and decreasing) of alcohol use based upon age at onset of drinking and drinking patterns [8, 13–15]. The "low" group consisted of individuals who either drank lightly or abstained completely across the entire adolescent-adult lifespan. The "high" group consisted of adolescents who drank heavily at a late-onset or chronically across the adolescent-adult lifespan. The



[©] Springer Nature Switzerland AG 2021

K. H. Todd et al. (eds.), Oncologic Emergency Medicine, https://doi.org/10.1007/978-3-030-67123-5_9

Department of Emergency Medicine, Yale New Haven Hospital, Yale School of Medicine, New Haven, CT, USA e-mail: edouard.coupet@yale.edu

"increasing" group consisted of adolescents who began drinking heavily in their late adolescence or emerging adult years. Shortly after this developmental period, their alcohol use declined. The final group of "decreasing" individuals began drinking heavily early in their adolescence and declined as they transitioned into late adolescence.

In 2018, results from the National Survey on Drug Use and Health (NSDUH) showed that nearly 140 million Americans aged 12 or older drank and 67 million reported binge drinking alcohol in the past month. Among adolescents aged 12–17, 4.7% reported binge drinking in the past month. Among young adults aged 18–25 and adults aged 26 and older, 34.9% and 25.1% reported binge drinking alcohol in the past month, respectively. Approximately 15 million Americans aged 12 or older meet the criteria for AUD as defined by the *Diagnostic and Statistical Manual of Mental Disorders IV (DSM-IV)* criteria for abuse or dependence. An estimated 1.6% of adolescents aged 12–17 meet the criteria for AUD. Among adults, 10.1% of those aged 18–25 and 5.1% of those aged 26 and older meet the criteria for AUD [16].

A variety of public health problems are either directly or indirectly caused by excessive alcohol consumption, including hepatic and pancreatic diseases, cancers, diabetes, suicide, cardiovascular disease, infectious diseases, and both intentional and unintentional injuries [17]. Regulation of alcohol sales and taxes on alcoholic beverages has been instituted at both federal and state levels to mitigate excessive alcohol consumption and alcohol-related disease. There is strong evidence to support these efforts [18-23]. One study determined that alcoholic beverage tax increases in 1983 and 2002 in Alaska which reduced the rate of death from alcoholrelated diseases by 29% and 11%, respectively [24]. A systematic review of literature that evaluated the effect of alcohol purchasing hours on several alcohol-related harms determined that regulation of these hours can reduce alcoholrelated hospitalizations and injuries, such as motor vehicle collisions (MVCs) [25].

Research has informed a spectrum of drinking levels. According to the 2015–2020 Dietary Guidelines for Americans, a standard drink is defined as 1.5 oz (45 mL) of 80-proof spirits, 12 oz of beer, or 5 oz of table wine. Each of these contains approximately 14.5 g of absolute ethanol. For those of legal drinking age, moderate alcohol consumption is defined as one drink per day for women and up to two drinks per day for men. Binge drinking is defined as a pattern of alcohol consumption that increases the blood alcohol concentration (BAC) to 0.08% or greater. This usually happens after four drinks for women and five drinks for men within 2 h. Heavy alcohol use is defined as greater than four drinks a day for women. Both binge and heavy alcohol use increase an individual's risk of adverse and negative consequences [26].

In this chapter, we will review common emergency department (ED) encounters for alcohol-related diseases and injuries, the identification and diagnosis of AUD, the ED approach to prevention of AUD, and the role of ED-based alcohol screening, brief intervention, and referral to treatment (SBIRT) in preventing alcohol related-cancers.

ED Encounters for Alcohol-Related Disease and Injury

Acute Conditions Related to Alcohol Use

From 2006 to 2014, the number of alcohol-related ED encounters in the USA increased by nearly 62% [27]. The unhealthy consumption of alcohol is a substantial burden on EDs, causing both injury and disease. Injury is one of the most common causes of alcohol-related ED encounters, accounting for up to 50% of all alcohol-related ED encounters in one study [28]. Alcohol has a well-established link to many types of unintentional injuries, particularly because of its psychomotor impairing effects [6, 29]. A meta-analysis of 28 articles of acute alcohol consumption and injury demonstrated a strong dose-response relationship of acute alcohol consumption and both MVC and non-MVC injuries [30]. Moreover, drivers who drink alcohol, yet are not legally impaired, cause thousands of deaths as well [31]. In a multisite prospective cohort study of both intoxicated and nonintoxicated injured drivers, alcohol was found to be a significant predictor of morbidity post-injury [32]. Previous research shows that younger drivers who have consumed alcohol are at highest risk for a fatal MVC because of their lower alcohol tolerance and relative driving inexperience [24, 33, 34]. Prior consumption of alcohol is also highly associated with increased injury severity, longer hospitalization, and higher healthcare costs in bicycle-related injuries presenting to the ED [35].

Intentional injuries, which commonly present to the ED, have an even greater association with alcohol use [36–38]. Alcohol consumption is well-known to increase aggressive behaviors and decrease inhibition [39]. Existing literature shows a strong link between alcohol use and interpersonal violence [36–38, 40, 41]. In a meta-analysis of 37 EDs across 18 different countries, 44.1% assault-related injuries were attributed to alcohol use, the highest for all types of injuries [40]. Another ED-based study found higher rates of a positive BAC in those who suffered an assault-related injury compared to those who were injured due to other causes [42].

Although the relationship is not as strong as that for nonpartner violence, there is evidence to support the link between alcohol and intimate partner violence [43, 44]. A metaanalysis determined a small to moderate effect size for the link between alcohol use and male-to-female partner violence and a small effect size for female-to-male partner violence [43]. Previous literature also demonstrates an association between alcohol use and behavioral health disorders, which increase the risk for suicide [45, 46].

Chronic Conditions Related to Alcohol Use

In the USA, the number of ED encounters for chronic conditions related to alcohol use increased by nearly 76% between 2006 and 2014. Harmful alcohol use, as indicated by the World Health Organization (WHO), was associated with 48% of liver cirrhosis cases [39]. There is substantial evidence to support the deleterious effects of alcohol on liver function and its role in the development of liver diseases seen in the ED such as cirrhosis, hepatic encephalopathy, and hepatitis [47–49]. In a prospective study of middle-aged women in the UK, daily alcohol consumption, not paired with meals, was associated with a twofold increase in liver cirrhosis. Moreover, patients with alcohol-related liver disease who have refrained from alcohol use experienced an increase in their life expectancy [49]. Another prospective population-based study demonstrated that the relative risk of developing alcohol-related liver disease significantly increased when women drank 7-13 and men 14-27 alcoholic beverages per week [50]. Evidence shows that current recommendations for safe alcohol consumption thresholds may be too high, further contributing to the increased alcoholrelated disease burden and harm [51, 52].

Since the turn of the century, there is increasing evidence that supports the casual link between alcohol consumption and cancer [53-56]. Alcohol consumption is linked to cancers of the head and neck, liver, gastrointestinal (GI) tract, and for women, breast [57]. Although the mechanism by which it causes cancer is not well understood, alcohol is believed to have several different carcinogenic mechanisms which likely differ by organ site. Acetaldehyde, the primary toxic metabolite of alcohol, is believed to alter deoxyribonucleic acid (DNA) strands [53, 57, 58]. Alcohol also reduces the blood concentration of important antioxidants including vitamins A and E, folic acid, thiamine, zinc, and iron. Additionally, it suppresses the immune system and can potentiate the effects of other carcinogenic agents [53]. Fortunately, evidence suggests stopping alcohol consumption can reduce the risks of developing certain types of cancers [59].

Along with tobacco use, alcohol consumption is one of the largest risk factors for cancers of the head and neck [60]. Previous research demonstrates that alcohol accounts for 26.4% of lip and oral cancers, 30.5% of all pharyngeal cancers, 21.6% of laryngeal cancers, and 16.9% of esophageal cancers [61]. Evidence also supports a dose-dependent rela-

tionship between alcohol use and head and neck cancers. In one study, participants who drank more than seven alcoholic beverages per day had four times the risk of developing cancers of the head and neck compared to those who abstained from drinking [62]. A meta-analysis of alcohol use and risk of head and neck cancers demonstrated that individuals who drank lightly, moderately, and heavily had relative risks of developing cancers of the head and neck of 1.13, 1.83, and 5.13, respectively [63]. There is also evidence to suggest that individuals who continue to drink, with a primary head and neck cancer diagnosis, are at increased risk of both primary cancer recurrence and development of a second primary cancer. A multisite study of more than 4000 individuals with head and neck cancer showed that participants who drank more than one alcoholic beverage per day had an increased risk of developing a second primary cancer [64]. A singlesite retrospective study of individuals with a primary head and neck cancer diagnosis showed that consuming 8-14 drinks per week was associated with an elevated risk of recurrence.

Heavy alcohol use is among the strongest risk factors for developing hepatocellular carcinoma (HCC) [65, 66]. Alcohol is believed to be both directly responsible for HCC and indirectly by causing liver cirrhosis, a well-known risk factor for HCC [61]. Alcohol has been proposed to cause oxidative stress, leading to the generation of free oxygen radicals and consequently, DNA damage [61]. A previous epidemiological study suggests that drinking more than 80 g/ day for at least 10 years substantially increases the risk of developing HCC [67]. Evidence also shows a synergistic effect when heavy alcohol consumption is paired with the hepatitis B or C virus infection. A case-control study of 115 patients with HCC demonstrated that those who report heavy alcohol use and have chronic hepatitis had an odds ratio (OR) of 53.9, compared to an OR of 2.4 with alcohol alone or 19.1 with the virus alone [68].

Although the evidence may not be as strong as that for other malignancies, there is evidence to support an association between alcohol use and cancer of parts of the GI tract, notably the esophagus, colon, and rectum. Up to 75% of cases of esophageal cancer can be attributed to alcohol use. Chronic alcohol consumption is understood to increase the esophageal mucosa's susceptibility to carcinogens [61]. Previous research shows a strong link between esophageal squamous cell carcinoma and alcohol consumption [69]. However, research does not support the same relationship for gastric cancer. A review of over 40 epidemiological studies of gastric cancer and chronic alcohol use did not find any association between the two [70]. Numerous studies suggest an increasing dose-dependent relationship between alcohol use and colorectal cancer [71]. A review of over 50 cohort and case-control studies found increased association between alcohol use and colorectal cancer by a factor of 2 [72].

The International Agency for Research on Cancer determined alcohol consumption to have a causal relationship to breast cancer, which exists in both premenopausal and postmenopausal women [69]. Existing literature suggests a 7-10% increase in risk for breast cancer for every 10 g (~1 drink) of alcohol/day consumed by an adult woman. This relationship has even been found in women who drink alcohol lightly [73-76]. A proposed mechanism behind this relationship appears to be the increased estrogen levels found in women who drink [75]. Alcohol has also been linked to recurrence of breast cancer and increased mortality from the disease. In a review, having more than 6 g of alcohol/day was associated with an increase in recurrence of and death from breast cancer. This relationship was determined to be even stronger in postmenopausal and obese women [77].

Special Considerations for Health Disparities

Existing data of alcohol consumption patterns demonstrate differences and disparities across ethnic and racial minority groups. According to the NSDUH, reported past month use of any alcohol in individuals aged 12 and older was highest in Whites (57.7%), followed by Blacks (43.6%), Latinos (43%), Native Hawaiians/Pacific Islanders (38.4%), American Indians/Alaska Native (37.3%), and Asians (34.5%). However, Native Hawaiians/ Pacific Islander populations were most likely to report heavy alcohol use in the past month (8.9%), while Asian populations reported the lowest rates (2%) [16]. White and Native American populations have the highest risk for developing AUD [78].

Once alcohol dependence develops, Black and Latino populations have the highest rates of recurrent or persistent dependence [79]. A review of 19 studies of racial/ethnic disparities and alcohol-attributable injury determined that Native Americans had the highest rate of alcohol-attributable injuries such as MVCs, self-injury, and falls compared to other racial/ethnic groups [80]. Black and Latino populations are also at greatest risk for developing alcohol-related liver disease [81]. Previous research suggests that gaps in alcohol treatment utilization are highest in Latino populations, particularly among those primarily Spanish-speaking [82–85]. Evidence demonstrates that alcohol treatment programs should take these socioeconomic and cultural factors into consideration when addressing disparities among racial/ethnic groups [86–89].

Identification and Diagnosis of Alcohol Use Disorder

Alcohol-Related Conditions

There were over 138 million ED encounters in the USA in 2017. Alcohol use contributed to a substantial portion of these encounters. Over 4.2 million ED encounters were related to alcohol misuse, abuse, or dependence [90]. Furthermore, acute alcohol-related disorders, such as injury, are among the leading causes of alcohol-related ED encounters. The list of all of the International Classification of Diseases, Tenth Revision (ICD-10), alcohol-related conditions an ED clinician may encounter is provided in Table 9.1.

Diagnosis of Alcohol Use Disorder

Diagnostic and Statistical Manual of Mental Disorders, Fifth Edition (DSM-5), criteria for diagnosis of AUD has shifted to a broader spectrum based on severity as compared to the dual model of alcohol abuse and dependence within the DSM-IV. In the DSM-IV, there were four diagnostic criteria for alcohol abuse and seven for dependence. In the DSM-5, the diagnostic criteria, which highlight craving, loss of control, tolerance, and withdrawal, have increased to 11 (Box 9.1). Individuals with two to three criteria meet the diagnosis of mild AUD; individuals with four to five, moderate; and individuals with six or more,

Box 9.1 *Diagnostic and Statistical Manual of Mental Disorders*, Fifth Edition (DSM-5): Alcohol Use Disorder

- 1. Alcohol-related failure to fulfill obligations
- 2. Recurrent drinking in hazardous situations
- 3. Continued drinking despite problems
- 4. Tolerance
- 5. Withdrawal
- 6. Drinking more or more often than intended
- 7. Unable to quit/cut back drinking
- 8. Spent a lot of time drinking/recovering
- 9. Gave up important activities due to drinking
- 10. Continued drinking despite consequences
- 11. Craving

Mild alcohol use disorder (AUD): 2–3 criteria required Moderate AUD: 4–5 criteria required Severe AUD: ≥6 criteria required

Table 9.1 ICD-10 codes for alcohol-related conditions

Group	ICD-10 code	Disorder
Acute conditions	F10.0	Mental and behavioral disorders due to use of alcohol, acute
		intoxication
	R78.0	Finding of alcohol in blood
	T51.0	Toxic effects of ethanol
	V12-V14 (0.3-0.9), V19.4-V19.6, V19.9, V20-	Road traffic injuries-non-pedestrian
	V28 (0.3–0.9), V29–V79 (0.4–0.9), V80.0–V86	
	(0.0–0.3), V87.0–V87.9, V89.2, V89.3, V89.9	
	V02–V04 (0.1,0.9), V06.1, V09.2, V09.3	Road traffic injuries-pedestrian
	W00-W19	Fall injuries
	W65-W74	Drowning
	W78–W79	Aspiration
	W24–W31, W45, W60	Occupational and machine injuries
	X00-X09	Fire injuries
	X78–X79, Y87.1	Assault
	X85–Y09, Y87.1	Child abuse
	X60–X84, Y87.0	Suicide
	X45	Accidental poisoning by and exposure to alcohol
	X65	Intentional self-poisoning by and exposure to alcohol
	Y15	Poisoning by and exposure to alcohol, undetermined intent
Chronic	C01–C06, C09–C10, C12–C14	Oropharyngeal cancer
conditions	C15	Esophageal cancer
	C22	Hepatocellular cancer
	C32	Laryngeal cancer
	C50	Female breast cancer
	E24.4	Alcohol-induced pseudo-Cushing's syndrome
	F10.1	Mental and behavioral disorders due to use of alcohol, harmful use
	F10.2	Dependence syndrome
	F10.3	Withdrawal state
	F10.4	Withdrawal state with delirium
	F10.5	Psychotic disorder
	F10.6	Amnesic syndrome
	F10.7	Residual and late-onset psychotic disorder
	F10.8	Other mental and behavioral disorders
	F10.9	Unspecified mental and behavioral disorder
	G31.2	Degeneration of nervous system due to alcohol
	G62.1	Alcohol polyneuropathy
	I42.6	Alcoholic cardiomyopathy
	K29.2	Alcoholic gastritis
	K70.0	Alcoholic fatty liver
	K70.1	Alcoholic hepatitis
	K70.2	Alcohol fibrosis and sclerosis of the liver
	K70.3	Alcoholic cirrhosis of the liver
	K/0.4	Alcoholic hepatic failure
	K70.9	Alcoholic liver disease, unspecified
	K85.2	Alcohol-induced acute pancreatitis
	K86.0	Alcohol-induced chronic pancreatitis
	003	Spontaneous abortion
	O36.5, P05, P07	Low birth weight

severe [91]. As a result of these changes in the DSM-5, some individuals may either have gained, lost, or changed in severity. The DSM-5 also removed alcohol-related legal problems and replaced it with the criteria of craving.

ED Approach to Prevention of Alcohol Use Disorder

Since many individuals seek emergency care for both acute and chronic alcohol-related concerns, the ED is the

ideal setting for the clinicians to approach and engage their patients regarding their alcohol use. With their patient's permission, ED clinicians should use this opportunity to discuss unhealthy alcohol use in an open, honest, nonconfrontational, nonjudgmental manner. To promote behavior change, ED clinicians should use health promotion, which is defined by WHO as "the process of enabling people to increase control over and to improve their health" [92]. For those who abstain or who are moderate alcohol consumers, this includes education about unhealthy/high-risk drinking and reinforcement of existing healthy behaviors. Individuals with high-risk drinking should be encouraged to reduce their drinking, ideally below the high-risk limits or less if possible. Federal dietary guidelines, endorsed by the National Institute on Alcohol Abuse and Alcoholism (NIAAA), clearly state that a person who does not drink alcohol should not start drinking for any reason [26].

Some individuals who report unhealthy or high-risk drinking may not be prepared to abstain from alcohol use completely. As an alternative to promoting abstinence, ED clinicians should provide harm reduction education. Harm reduction is a strategy to minimize the acute harms associated with alcohol intoxication [93–95]. It includes reducing alcohol consumption, avoiding drinking and driving, and changing drinking patterns. A systematic review of 63 studies of reduced alcohol-related injuries, slow progression of alcohol-related diseases such as cirrhosis and cardiomy-opathy, and reduce withdrawal symptoms [96].

A substantial amount of literature has evaluated screening and providing an ED-based intervention for unhealthy alcohol use. Over two decades ago, a framework for intervention known as screening, brief intervention, and referral to treatment (SBIRT) was developed and tested in both acute and primary care settings for high-risk behaviors such as substance use. SBIRT involves a psychosocial intervention which utilizes principles of motivational interviewing to encourage individuals with unhealthy alcohol use to pursue treatment. The intervention, more specifically the brief negotiated interview (BNI), relies upon a patient-centered discussion that assesses their willingness to reduce their alcohol use toward healthier limits. It utilizes principles of both harm reduction and health promotion. The American College of Emergency Physicians, the Committee on Trauma of the American College of Surgeons, and the Emergency Nurses Association have all recommended SBIRT for addressing unhealthy alcohol use in their respective clinical settings [97-99].

There is modest evidence to support the effectiveness of ED alcohol SBIRT in reducing alcohol use and alcohol-related consequences in the ED [93, 100–103]. A 2007 sys-

tematic review of 13 studies found SBIRT did not have any effect on quantity/frequency of drinking at 12 months and was inconclusive at 3 months. However, it did find a 41% reduction in the odds of alcohol-related injury at 6 and 12 months following the initial ED encounter [104]. A more recent systematic review of 35 articles that evaluated the effect of a brief intervention on ED patients who were screened as high-risk for AUD showed a short-term effect in reducing alcohol consumption in those who drank low or moderately [101]. Previous research also shows that ED SBIRT for alcohol use is cost-effective [105].

The BNI of SBIRT has been developed and tested in numerous different modalities including in-person, computerized, and by smartphone (Box 9.2). A previous concern about SBIRT was that characteristics of the BNI and patient had not been evaluated. A 2019 clinical trial evaluated 750 ED patients with unhealthy alcohol use and randomly assigned them to receive either a computer-delivered BNI, therapist-delivered BNI with computer guidance, or enhanced usual care. The main outcome was alcohol use at 3, 6, and 12 months. Moderation of the intervention effect was also tested by gender, age, and severity of alcohol disorder. Overall, there was no difference in the main effects of either computer-delivered or therapist-delivered BNI compared to enhanced usual care. However, the therapist-delivered BNI group was more effective among patients with moderate to severe drinking patterns, while the computer-delivered BNI was more effective among younger participants [106]. Other benefits to computer-delivered SBIRT include its use in resource-limited EDs. There is also a reduction in time burden for care providers, enhanced fidelity of intervention delivery, and the potential for multilingual administration [86, 107, 108]. Overall, alcohol ED SBIRT represents a critical opportunity to identify patients with untreated AUD, particularly those who may not be seeking treatment. Once identified, these patients can receive a brief intervention or be linked to specialized treatment. Without this service, this precious opportunity is missed. Furthermore, both the NIAAA and Public Health Task Force recommend screening for unhealthy alcohol use; there appears to be no harm to the patients in doing so [109, 110].

Box 9.2 Modes of ED SBIRT for Alcohol Use Disorder

- Face-to-face physician/nurse interaction
- · Specialty-trained paraprofessionals
- Computerized interaction
- Computerized and human interaction
- Smartphone-based interaction

Potential Value of ED-Based Alcohol SBIRT for Alcohol-Related Cancers

The ED serves a unique opportunity to detect unhealthy alcohol use and render services to those who present with concerns related to at-risk and hazardous alcohol use. In addition to decreasing use, ED-based SBIRT has shown some promise in reducing alcohol-related consequences such as MVCs [101, 103, 104, 111]. However, to date, there have been no studies that evaluate the effect of ED SBIRT on alcohol-attributable cancers. Existing evidence shows that alcohol use has a strong link to several types of cancers including head and neck, liver, GI tract, and breast [55, 57, 60, 62, 63, 72–74, 112]. By decreasing alcohol use, ED SBIRT has the potential to reduce exposure to alcohol (i.e., physiologically) and the risks of developing alcohol-attributable malignancy and other diseases associated with it, such as cirrhosis. Previous evidence has also shown that ED SBIRT can increase treatment access, particularly for those with alcohol dependence and AUD. A 2015 systematic review found that 7 out of 15 studies evaluated whether or not participants pursued treatment. Four studies compared the referral to treatment rates in the group that received BNI to the control group. One study showed a higher referral to treatment rate, in favor of the group that received BNI [113].

Recognition and treatment of severe AUD and its sequelae in the ED are also important among those with terminally ill disease such as advanced-stage cancer and liver cirrhosis. Alcohol dependency affects up to 28% of palliative care inpatients [114]. As high as 87% of patients with severe AUD receiving palliative care would not have been diagnosed had they not been screened [115]. ED clinicians should consider ongoing or worsening AUD in their differential diagnosis, particularly when evaluating patients with advanced-stage cancer. They should also consider identifying what appears to be signs and symptoms of acute alcohol withdrawal albeit in a less familiar context. Although evidence of the management of alcohol dependency among terminally ill patients is limited, ED clinicians should be proactive in recognizing and treating symptoms of acute alcohol withdrawal in this population. Low-dose benzodiazepines, such as lorazepam, are recommended with careful attention to those with liver impairment. Terminally ill patients who are interested in receiving detoxification are most appropriate for referral to the inpatient setting [116]. Alcohol ED SBIRT may also provide additional benefits in identifying unhealthy alcohol use in patients who have either received a new diagnosis of cancer or may be terminally ill.

Conclusion

EDs across the world diagnose and treat acute conditions related to unhealthy alcohol use such as MVCs and more chronic conditions such as liver cirrhosis and breast cancer. Thus, the ED is the ideal setting to identify patients with unhealthy alcohol use and motivate them to pursue treatment using SBIRT. It has shown promise in reducing alcohol consumption and alcohol-related injuries. By reducing alcohol consumption, ED SBIRT has the potential to reduce the risk of developing alcohol-attributable cancers as well. Tobacco and alcohol ED SBIRT, when used in conjunction, have the potential to have synergistic effects in reducing the overall burden of cancer [117]. However, additional research is needed to evaluate the effects of alcohol SBIRT in the ED on alcohol-attributable cancer.

References

- Ahern J, Galea S, Hubbard A, Midanik L, Syme SL. "Culture of drinking" and individual problems with alcohol use. Am J Epidemiol. 2008;167(9):1041–9.
- Measham F, Brain K. "Binge" drinking, British alcohol policy and the new culture of intoxication. Crime Media Cult. 2005;1:262–83.
- Rehm J, Mathers C, Popova S, Thavorncharoensap M, Teerawattananon Y, Patra J. Global burden of disease and injury and economic cost attributable to alcohol use and alcohol-use disorders. Lancet. 2009;373(9682):2223–33.
- GBD 2016 Alcohol Collaborators. Alcohol use and burden for 195 countries and territories, 1990–2016: a systematic analysis for the Global Burden of Disease Study 2016. Lancet. 2018;392(10152):1015–35. Erratum in: Lancet. 2018;392(10153):1116. Erratum in: Lancet. 2019;393(10190):e44.
- Rehm J, Anderson P, Barry J, Dimitrov P, Elekes Z, Feijão F, et al. Prevalence of and potential influencing factors for alcohol dependence in Europe. Eur Addict Res. 2015;21(1):6–18.
- Rehm J, Baliunas D, Borges GL, Graham K, Irving H, Kehoe T, et al. The relation between different dimensions of alcohol consumption and burden of disease: an overview. Addiction. 2010;105(5):817–43.
- Donovan JE. Adolescent alcohol initiation: a review of psychosocial risk factors. J Adolesc Health. 2004;35(6):529.e7–18.
- Biehl MC, Natsuaki MN, Ge X. The influence of pubertal timing on alcohol use and heavy drinking trajectories. J Youth Adolesc. 2007;36(2):153–67.
- 9. Windle M. Alcohol use among adolescents and young adults. Alcohol Res Health. 2003;27(1):79–85.
- Brown SA, Tapert SF. Adolescence and the trajectory of alcohol use: basic to clinical studies. Ann NY Acad Sci. 2004;1021:234–44.
- Johnston LD, Miech RA, O'Malley PM, Bachman JG, Schulenberg JE, Patrick ME. Monitoring the future national survey results on drug use, 1975–2018: overview, key findings on adolescent drug use. Ann Arbor: Institute for Social Research, University of Michigan; 2019. https://files.eric.ed.gov/fulltext/ED594190.pdf
- Grant BF. Age at smoking onset and its association with alcohol consumption and DSM-IV alcohol abuse and dependence: results

from the National Longitudinal Alcohol Epidemiologic Survey. J Subst Abus. 1998;10(1):59–73.

- Danielsson AK, Wennberg P, Tengström A, Romelsjö A. Adolescent alcohol use trajectories: predictors and subsequent problems. Addict Behav. 2010;35(9):848–52.
- Maggs JL, Schulenberg JE. Trajectories of alcohol use during the transition to adulthood. Alcohol Res Health. 2004;28(4):195–201.
- Sher KJ, Jackson KM, Steinley D. Alcohol use trajectories and the ubiquitous cat's cradle: cause for concern? J Abnorm Psychol. 2011;120(2):322–35. Erratum: J Abnorm Psychol. 2011;120(2):510.
- 16. Substance Abuse and Mental Health Services Administration. Key substance use and mental health indicators in the United States: results from the 2017 National Survey on Drug Use and Health (HHS Publication No. SMA 18-5068, NSDUH Series H-53). Rockville: Center for Behavioral Health Statistics and Quality, Substance Abuse and Mental Health Services Administration; 2018. https://www.samhsa.gov/data/
- 17. Rehm J. The risks associated with alcohol use and alcoholism. Alcohol Res Health. 2011;34(2):135–43.
- Elder RW, Lawrence B, Ferguson A, Naimi TS, Brewer RD, Chattopadhyay SK, Task Force on Community Preventive Services, et al. The effectiveness of tax policy interventions for reducing excessive alcohol consumption and related harms. Am J Prev Med. 2010;38(2):217–29.
- Nelson TF, Xuan Z, Babor TF, Brewer RD, Chaloupka FJ, Gruenewald PJ, et al. Efficacy and the strength of evidence of U.S. alcohol control policies. Am J Prev Med. 2013;45(1):19–28.
- Chaloupka FJ, Grossman M, Saffer H. The effects of price on alcohol consumption and alcohol-related problems. Alcohol Res Health. 2002;26(1):22–34.
- Purshouse RC, Meier PS, Brennan A, Taylor KB, Rafia R. Estimated effect of alcohol pricing policies on health and health economic outcomes in England: an epidemiological model. Lancet. 2010;375(9723):1355–64.
- 22. Middleton JC, Hahn RA, Kuzara JL, Elder R, Brewer R, Chattopadhyay S, Task Force on Community Preventive Services, et al. Effectiveness of policies maintaining or restricting days of alcohol sales on excessive alcohol consumption and related harms. Am J Prev Med. 2010;39(6):575–89.
- Fell JC. Approaches for reducing alcohol-impaired driving: evidence-based legislation, law enforcement strategies, sanctions, and alcohol-control policies. Forensic Sci Rev. 2019;31(2):161–84.
- 24. Wagenaar AC, Maldonado-Molina MM, Wagenaar BH. Effects of alcohol tax increases on alcohol-related disease mortality in Alaska: time-series analyses from 1976 to 2004. Am J Public Health. 2009;99(8):1464–70. Erratum in: Am J Public Health. 2012;102(4):584.
- Sanchez-Ramirez DC, Voaklander D. The impact of policies regulating alcohol trading hours and days on specific alcohol-related harms: a systematic review. Inj Prev. 2018;24(1):94–100.
- U.S. Department of Health and Human Services and U.S. Department of Agriculture. 2015–2020 Dietary Guidelines for Americans. 8th ed. 2015. http://health.gov/dietaryguidelines/2015/guidelines.
- 27. White AM, Slater ME, Ng G, Hingson R, Breslow R. Trends in alcohol-related emergency department visits in the United States: results from the nationwide emergency department sample, 2006 to 2014. Alcohol Clin Exp Res. 2018;42(2):352–9.
- Li G, Keyl PM, Rothman R, Chanmugam A, Kelen GD. Epidemiology of alcohol-related emergency department visits. Acad Emerg Med. 1998;5(8):788–95.
- Culhane J, Silverglate B, Freeman C. Alcohol is a predictor of mortality in motor vehicle collisions. J Saf Res. 2019;71:201–5.
- 30. Taylor B, Irving HM, Kanteres F, Room R, Borges G, Cherpitel C, et al. The more you drink, the harder you fall: a systematic review and meta-analysis of how acute alcohol consumption and

injury or collision risk increase together. Drug Alcohol Depend. 2010;110(1–2):108–16.

- Lira MC, Sarda V, Heeen TC, Miller M, Naimi TS. Alcohol policies and motor vehicle crash deaths involving blood alcohol concentrations below 0.08. Am J Prev Med. 2020;58(5):622–9.
- Shih HC, Hu SC, Yang CC, Ko TJ, Wu JK, Lee CH. Alcohol intoxication increases morbidity in drivers involved in motor vehicle accidents. Am J Emerg Med. 2003;21(2):91–4.
- 33. Wagenaar AC. Alcohol, young drivers, and traffic accidents. Lexington: Lexington Books; 1983.
- Wagenaar AC, O'Malley PM, LaFond C. Lowered legal blood alcohol limits for young drivers: effects on drinking, driving, and driving-after-drinking behaviors in 30 states. Am J Public Health. 2001;91(5):801–4.
- 35. Spaite DW, Criss EA, Weist DJ, Valenzuela TD, Judkins D, Meislin HW. A prospective investigation of the impact of alcohol consumption on helmet use, injury severity, medical resource utilization, and health care costs in bicycle-related trauma. J Trauma. 1995;38(2):287–90.
- Cherpitel CJ. Alcohol and injuries resulting from violence: a review of emergency room studies. Addiction. 1994;89(2):157–65.
- Cherpitel CJ. Alcohol and violence-related injuries in the emergency room. Recent Dev Alcohol. 1997;13:105–18.
- Vinson DC, Borges G, Cherpitel CJ. The risk of intentional injury with acute and chronic alcohol exposures: a case-control and casecrossover study. J Stud Alcohol. 2003;64(3):350–7.
- 39. World Health Organization. Global status report on alcohol and health 2018. Geneva: World Health Organization; 2018.
- 40. Cherpitel CJ, Ye Y, Bond J, Borges G, Monteiro M, Chou P, Hao W. Alcohol attributable fraction for injury morbidity from the dose-response relationship of acute alcohol consumption: emergency department data from 18 countries. Addiction. 2015;110(11):1724–32.
- Macdonald S, Cherpitel CJ, Borges G, Desouza A, Giesbrecht N, Stockwell T. The criteria for causation of alcohol in violent injuries based on emergency room data from six countries. Addict Behav. 2005;30(1):103–13.
- Cherpitel CJ, Ye Y, Bond J, Room R, Borges G. Attribution of alcohol to violence-related injury: self and other's drinking in the event. J Stud Alcohol Drugs. 2012;73(2):277–84.
- Foran HM, O'Leary KD. Alcohol and intimate partner violence: a meta-analytic review. Clin Psychol Rev. 2008;28(7):1222–34.
- 44. Leonard KE. Alcohol and intimate partner violence: when can we say that heavy drinking is a contributing cause of violence? Addiction. 2005;100(4):422–5.
- Pompili M, Serafini G, Innamorati M, Dominici G, Ferracuti S, Kotzalidis GD, et al. Suicidal behavior and alcohol abuse. Int J Environ Res Public Health. 2010;7(4):1392–431.
- 46. Borges G, Loera CR. Alcohol and drug use in suicidal behaviour. Curr Opin Psychiatry. 2010;23(3):195–204.
- Hydes T, Gilmore W, Sheron N, Gilmore I. Treating alcoholrelated liver disease from a public health perspective. J Hepatol. 2019;70(2):223–36.
- Wong RJ, Gish RG. Clinical epidemiology of chronic liver diseases. Cham: Springer International; 2019.
- 49. Simpson RF, Hermon C, Liu B, Green J, Reeves GK, Beral V, Floud S, Million Women Study Collaborators. Alcohol drinking patterns and liver cirrhosis risk: analysis of the prospective UK Million Women Study. Lancet Public Health. 2019;4(1):e41–8.
- Becker U, Deis A, Sørensen TI, Grønbaek M, Borch-Johnsen K, Müller CF, et al. Prediction of risk of liver disease by alcohol intake, sex, and age: a prospective population study. Hepatology. 1996;23(5):1025–9.
- 51. Wood AM, Kaptoge S, Butterworth AS, Willeit P, Warnakula S, Bolton T, Paige E, Emerging Risk Factors Collaboration/EPIC-CVD/UK Biobank Alcohol Study Group, et al. Risk thresholds for

alcohol consumption: combined analysis of individual-participant data for 599 912 current drinkers in 83 prospective studies. Lancet. 2018;391(10129):1513–23. Erratum in: Lancet. 2018;391(10136):2212.

- Connor J, Hall W. Thresholds for safer alcohol use might need lowering. Lancet. 2018;391(10129):1460–1.
- Boffetta P, Hashibe M. Alcohol and cancer. Lancet Oncol. 2006;7(2):149–56.
- Bosque-Prous M, Mendieta-Paredes J, Bartroli M, Brugal MT, Espelt A. Cancer and alcohol consumption in people aged 50 years or more in Europe. Alcohol Alcohol. 2018;53(3):317–24.
- Choi YJ, Myung SK, Lee JH. Light alcohol drinking and risk of cancer: a meta-analysis of cohort studies. Cancer Res Treat. 2018;50(2):474–87.
- Druesne-Pecollo N, Tehard B, Mallet Y, Gerber M, Norat T, Hercberg S, Latino-Martel P. Alcohol and genetic polymorphisms: effect on risk of alcohol-related cancer. Lancet Oncol. 2009;10(2):173–80.
- 57. Rehm J, Soerjomataram I, Ferreira-Borges C, Shield KD. Does alcohol use affect cancer risk? Curr Nutr Rep. 2019;8(3):222–9.
- Vaca CE, Nilsson JA, Fang JL, Grafström RC. Formation of DNA adducts in human buccal epithelial cells exposed to acetaldehyde and methylglyoxal in vitro. Chem Biol Interact. 1998;108(3):197–208.
- Connor J. Alcohol consumption as a cause of cancer. Addiction. 2017;112(2):222–8. https://doi.org/10.1111/add.13477. Epub 2016 Jul 21. PMID: 27442501.
- 60. Hashibe M, Brennan P, Chuang SC, Boccia S, Castellsague X, Chen C, et al. Interaction between tobacco and alcohol use and the risk of head and neck cancer: pooled analysis in the International Head and Neck Cancer Epidemiology Consortium. Cancer Epidemiol Biomark Prev. 2009;18(2):541–50.
- 61. Testino G. The burden of cancer attributable to alcohol consumption. Maedica (Bucur). 2011;6(4):313–20.
- Longnecker MP, Enger SM. Epidemiologic data on alcoholic beverage consumption and risk of cancer. Clin Chim Acta. 1996;246(1–2):121–41.
- Bagnardi V, Rota M, Botteri E, Tramacere I, Islami F, Fedirko V, et al. Alcohol consumption and site-specific cancer risk: a comprehensive dose-response meta-analysis. Br J Cancer. 2015;112(3):580–93.
- 64. Leoncini E, Vukovic V, Cadoni G, Giraldi L, Pastorino R, Arzani D, et al. Tumour stage and gender predict recurrence and second primary malignancies in head and neck cancer: a multicentre study within the INHANCE consortium. Eur J Epidemiol. 2018;33(12):1205–18.
- Morgan TR, Mandayam S, Jamal MM. Alcohol and hepatocellular carcinoma. Gastroenterology. 2004;127(5 Suppl 1):S87–96.
- El-Serag HB, Mason AC. Rising incidence of hepatocellular carcinoma in the United States. N Engl J Med. 1999;340(10):745–50.
- 67. Donato F, Tagger A, Gelatti U, Parrinello G, Boffetta P, Albertini A, et al. Alcohol and hepatocellular carcinoma: the effect of life-time intake and hepatitis virus infections in men and women. Am J Epidemiol. 2002;155(4):323–31.
- Hassan MM, Hwang LY, Hatten CJ, Swaim M, Li D, Abbruzzese JL, Beasley P, Patt YZ. Risk factors for hepatocellular carcinoma: synergism of alcohol with viral hepatitis and diabetes mellitus. Hepatology. 2002;36(5):1206–13.
- IARC Working Group on the Evaluation of Carcinogenic Risks to Humans. Alcohol consumption and ethyl carbamate. IARC Monogr Eval Carcinog Risks Hum. 2010;96:3-1383.
- Franke A, Teyssen S, Singer MV. Alcohol-related diseases of the esophagus and stomach. Dig Dis. 2005;23(3–4):204–13.
- Seitz HK, Stickel F. Molecular mechanisms of alcohol-mediated carcinogenesis. Nat Rev Cancer. 2007;7(8):599–612.
- 72. Franceschi S, La Vecchia C. Alcohol and the risk of cancers of the stomach and colon-rectum. Dig Dis. 1994;12(5):276–89.

- Brooks PJ, Zakhari S. Moderate alcohol consumption and breast cancer in women: from epidemiology to mechanisms and interventions. Alcohol Clin Exp Res. 2013;37(1):23–30.
- 74. Chen WY, Rosner B, Hankinson SE, Colditz GA, Willett WC. Moderate alcohol consumption during adult life, drinking patterns, and breast cancer risk. JAMA. 2011;306(17):1884–90.
- Key J, Hodgson S, Omar RZ, Jensen TK, Thompson SG, Boobis AR, et al. Meta-analysis of studies of alcohol and breast cancer with consideration of the methodological issues. Cancer Causes Control. 2006;17(6):759–70.
- 76. Smith-Warner SA, Spiegelman D, Yaun SS, van den Brandt PA, Folsom AR, Goldbohm RA, et al. Alcohol and breast cancer in women: a pooled analysis of cohort studies. JAMA. 1998;279(7):535–40.
- Dumitrescu RG, Shields PG. The etiology of alcohol-induced breast cancer. Alcohol. 2005;35(3):213–25.
- National Institute on Alcohol Abuse and Alcoholism. National Institutes of Health. Alcohol Alert. National Epidemiologic Survey on Alcohol and Related Conditions. No 70. Oct 2006. https://pubs.niaaa.nih.gov/publications/aa70/aa70.htm. Accessed 20 Aug 2020.
- Dawson DA, Grant BF, Stinson FS, Chou PS, Huang B, Ruan WJ. Recovery from DSM-IV alcohol dependence: United States, 2001-2002. Addiction. 2005;100(3):281–92.
- Keyes KM, Liu XC, Cerda M. The role of race/ethnicity in alcohol-attributable injury in the United States. Epidemiol Rev. 2012;34(1):89–102.
- Flores YN, Yee HF Jr, Leng M, Escarce JJ, Bastani R, Salmerón J, Morales LS. Risk factors for chronic liver disease in Blacks, Mexican Americans, and Whites in the United States: results from NHANES IV, 1999-2004. Am J Gastroenterol. 2008;103(9):2231–8.
- Schmidt L, Greenfield T, Mulia N. Unequal treatment: racial and ethnic disparities in alcoholism treatment services. Alcohol Res Health. 2006;29(1):49–54.
- Schmidt LA, Ye Y, Greenfield TK, Bond J. Ethnic disparities in clinical severity and services for alcohol problems: results from the National Alcohol Survey. Alcohol Clin Exp Res. 2007;31(1):48–56.
- Zemore SE, Mulia N, Ye Y, Borges G, Greenfield TK. Gender, acculturation, and other barriers to alcohol treatment utilization among Latinos in three National Alcohol Surveys. J Subst Abus Treat. 2009;36(4):446–56.
- 85. Bluthenthal RN, Jacobson JO, Robinson PL. Are racial disparities in alcohol treatment completion associated with racial differences in treatment modality entry? Comparison of outpatient treatment and residential treatment in Los Angeles County, 1998 to 2000. Alcohol Clin Exp Res. 2007;31(11):1920–6.
- Abujarad F, Vaca FE. m Health tool for alcohol use disorders among Latinos in emergency department. Proc Int Symp Hum Factors Ergon Healthc. 2015;4(1):12–9.
- Lotfipour S, Cisneros V, Anderson CL, Roumani S, Hoonpongsimanont W, Weiss J, et al. Assessment of alcohol use patterns among Spanish-speaking patients. Subst Abus. 2013;34(2):155–61.
- Arroyo JA, Miller WR, Tonigan JS. The influence of Hispanic ethnicity on long-term outcome in three alcohol-treatment modalities. J Stud Alcohol. 2003;64(1):98–104.
- Jones-Webb R, Snowden L, Herd D, Short B, Hannan P. Alcoholrelated problems among black, Hispanic and white men: the contribution of neighborhood poverty. J Stud Alcohol. 1997;58(5):539–45.
- Rui P, Kang K. National Hospital Ambulatory Medical Care Survey: 2017 emergency department summary tables. National Center for Health Statistics. Available from: https://www.cdc.gov/nchs/data/ nhamcs/web_tables/2017_ed_web_tables-508.pdf.
- American Psychiatric Association. Diagnostic and statistical manual of mental disorders (DSM-5®). 5th ed. Arlington: American Psychiatric Association; 2013.

- 92. World Health Organization. Regional Office for Europe. Health promotion: a discussion document on the concept and principles. Summary Report of the Working Group on Concept and Principles of Health Promotion. Copenhagen, 9–13 July 1984.
- D'Onofrio G, Fiellin DA, Pantalon MV, Chawarski MC, Owens PH, Degutis LC, et al. A brief intervention reduces hazardous and harmful drinking in emergency department patients. Ann Emerg Med. 2012;60(2):181–92.
- 94. Roche AM, Evans KR, Stanton WR. Harm reduction: roads less travelled to the holy grail. Addiction. 1997;92(9):1207–12.
- Wodak A. What is this thing called harm reduction? Int J Drug Policy. 1999;10:169–71.
- Charlet K, Heinz A. Harm reduction-a systematic review on effects of alcohol reduction on physical and mental symptoms. Addict Biol. 2017;22(5):1119–59.
- American College of Emergency Physicians. Alcohol screening in the emergency department. Ann Emerg Med. 2005;46(2):214–5.
- London JA, Dunn CW, Utter GH. Alcohol-related brief interventions as a criterion for American College of Surgeons Level I Trauma Center verification: how best to train the interventionists? J Trauma. 2011;70(4):931–8.
- 99. Strobbe S, Perhats C, Broyles LM. Expanded roles and responsibilities for nurses in screening, brief intervention, and referral to treatment (SBIRT) for alcohol use. J Addict Nurs. 2013;24(3):203–4.
- 100. D'Onofrio G, Degutis LC. Integrating project ASSERT: a screening, intervention, and referral to treatment program for unhealthy alcohol and drug use into an urban emergency department. Acad Emerg Med. 2010;17(8):903–11.
- 101. Barata IA, Shandro JR, Montgomery M, Polansky R, Sachs CJ, Duber HC, et al. Effectiveness of SBIRT for alcohol use disorders in the emergency department: a systematic review. West J Emerg Med. 2017;18(6):1143–52.
- 102. Biroscak BJ, Pantalon MV, Dziura JD, Hersey DP, Vaca FE. Use of non-face-to-face modalities for emergency department screening, brief intervention, and referral to treatment (ED-SBIRT) for high-risk alcohol use: a scoping review. Subst Abus. 2019;40(1):20–32.
- 103. Bruguera P, Barrio P, Oliveras C, Braddick F, Gavotti C, Bruguera C, et al. Effectiveness of a specialized brief intervention for at-risk drinkers in an emergency department: short-term results of a randomized controlled trial. Acad Emerg Med. 2018;25(5):517–25.
- 104. Havard A, Shakeshaft A, Sanson-Fisher R. Systematic review and meta-analyses of strategies targeting alcohol problems in emergency departments: interventions reduce alcohol-related injuries. Addiction. 2008;103:368–76.
- 105. Barbosa C, Cowell A, Bray J, Aldridge A. The cost-effectiveness of alcohol screening, brief intervention, and referral to treatment

(SBIRT) in emergency and outpatient medical settings. J Subst Abus Treat. 2015;53:1–8.

- 106. Fernandez AC, Waller R, Walton MA, Bonar EE, Ignacio RV, Chermack ST, et al. Alcohol use severity and age moderate the effects of brief interventions in an emergency department randomized controlled trial. Drug Alcohol Depend. 2019;194:386–94.
- 107. Vaca F, Winn D, Anderson C, Kim D, Arcila M. Feasibility of emergency department bilingual computerized alcohol screening, brief intervention, and referral to treatment. Subst Abus. 2010;31(4):264–9.
- Vaca FE, Winn D. The basics of alcohol screening, brief intervention and referral to treatment in the emergency department. West J Emerg Med. 2007;8(3):88–92.
- 109. National Institute on Alcohol Abuse and Alcoholism. A pocket guide for alcohol screening and brief intervention. Rockville: NIAAA; 2005. https://medicine.tulane.edu/sites/medicine.tulane. edu/files/pictures/Substance%20Use%20NIAA%20Alcohol%20 Use%20Pocket%20Guide.pdf.
- 110. US Preventive Services Task Force, Curry SJ, Krist AH, Owens DK, Barry MJ, Caughey AB, Davidson KW, et al. Screening and behavioral counseling interventions to reduce unhealthy alcohol use in adolescents and adults: US Preventive Services Task Force recommendation statement. JAMA. 2018;320(18):1899–909.
- 111. Mello MJ, Nirenberg TD, Longabaugh R, Woolard R, Minugh A, Becker B, Baird J, Stein L. Emergency department brief motivational interventions for alcohol with motor vehicle crash patients. Ann Emerg Med. 2005;45(6):620–5.
- 112. Marziliano A, Teckie S, Diefenbach MA. Alcohol-related head and neck cancer: summary of the literature. Head Neck. 2020;42(4):732–8.
- 113. Diestelkamp S, Drechsel M, Baldus C, Wartberg L, Arnaud N, Thomasius R. Brief in person interventions for adolescents and young adults following alcohol-related events in emergency care: a systematic review and European evidence synthesis. Eur Addict Res. 2016;22(1):17–35.
- 114. Hearne R, Connolly A, Sheehan J. Alcohol abuse: prevalence and detection in a general hospital. J R Soc Med. 2002;95(2):84–7.
- 115. Bruera E, Moyano J, Seifert L, Fainsinger RL, Hanson J, Suarez-Almazor M. The frequency of alcoholism among patients with pain due to terminal cancer. J Pain Symptom Manag. 1995;10(8):599–603.
- 116. MacCormac A. Alcohol dependence in palliative care: a review of the current literature. J Palliat Care. 2017;32(3–4):108–12.
- 117. Cunningham RM, Bernstein SL, Walton M, Broderick K, Vaca FE, Woolard R, Bernstein E, Blow F, D'Onofrio G. Alcohol, tobacco, and other drugs: future directions for screening and intervention in the emergency department. Acad Emerg Med. 2009;16(11):1078–88.

.

Richard T. Griffey

Ionizing Radiation

Case Study

A 28-year-old female with Crohn's disease presents with diffuse abdominal pain and low-grade fevers. She has had small-volume blood-tinged loose stools but no melena or vomiting. She was diagnosed with Crohn's disease at age 21 and has had several abdominopelvic CTs for staging and acute evaluations. She feels this may be a typical Crohn's flare but can't be sure. Knowing that as a young female at risk for repeat and multiple imaging with CT, which compounds an increased risk for luminal malignancies from her disease alone, the emergency physician discusses her care with her gastroenterologist and opts for imaging with magnetic resonance enterography (MRE) and admission, avoiding further ionizing radiation for non-obstructive symptoms.

Introduction

CT Is a Transformative Tool in Medicine

Computed tomography (CT) has numerous benefits that impact emergency care, including, but not limited to, decreasing negative appendectomy [1–5] and exploratory laparotomy rates, decreasing the need for hospitalization [6], allowing for safe discharge after exclusion of coronary disease [7], increasing provider [8] and patient [9, 10] confidence in diagnoses, and possibly even decreasing mortality [11] and increasing life expectancy [12]. Indeed, CT is fast, easy to obtain, relatively inexpensive, widely available, and highly sensitive, resulting in relatively low radiation exposure for the benefits obtained. The benefits of a CT *with appropriate indications* far outweigh the risks.

The Explosion in CT Utilization

In light of this, it is perhaps not surprising that CT use has skyrocketed over the last two decades, with a growth rate of 14% per year for about a 12-year period [13]. In 1981, 3 million CTs were performed, increasing to over 67 million exams by 2006 (Fig. 10.1) [14]. One in ten Americans undergoes a CT scan every year, and many undergo more than one [15].

Increasing awareness about cumulative radiation in addition to other policies and guidelines may have contributed to a plateauing of CT imaging rates observed in the years 2008– 2010 onward [16, 17]. Though this trend is true for overall CT rates, utilization in the ED has continued to increase [18]. It is estimated that one in seven ED patients undergoes CT and that 25% of CTs in the USA are performed in the ED [7]. In one study, 70% of the nearly one million non-elderly adults underwent at least one imaging study that included ionizing radiation, resulting in mean effective doses that nearly doubled the cumulative radiation expected from natural sources alone [19]. Similar patterns of CT use have been observed in pediatric populations as well, though more so in non-pediatric EDs [20, 21].

CT Contribution to Cumulative Radiation

Because of the much higher exposures that are imparted by CT when compared to radiographs, discussion of this modality drives the discussion about concerns over cumulative radiation exposure as well as costs [22]. Though by volume, radiographs comprise the majority of imaging studies, on a population basis, these account for a relatively small amount of cumulative radiation exposure [23]. Interventional diagnostic studies and therapeutic procedures, such as thallium scans and radiotherapy, can impart much higher radiation doses than CT but are not as commonly performed. In 2006, CT comprised about 17% of imaging procedures but was the source for over half the medical radiation dose in the USA (Fig. 10.2) [1]. The delayed nature of the carcinogenic effects



10

R. T. Griffey (⊠)

Department of Emergency Medicine, Department of Medicine, Barnes Jewish Hospital/Washington University School of Medicine, St. Louis, MO, USA e-mail: griffeyr@wustl.edu

K. H. Todd et al. (eds.), Oncologic Emergency Medicine, https://doi.org/10.1007/978-3-030-67123-5_10



Fig. 10.1 Growth of CT from 1980 to 2006 as a contributor to cumulative radiation exposure. (From the National Council on Radiation Protection and Measurements, https://ncrponline.org/publications/, with permission)



of radiation exposure makes overutilization an insidious problem that fails to signal the usual alarms among patients or providers. Of concern, cumulative radiation from diagnostic imaging is projected to account for up to 5% of future cancers in the USA [15, 24, 25].

Reasons for Increased CT Utilization

Many reasons for the observed increases have been proposed, including but not limited to an aging population, the wide availability of CT, the replacement of older (X-rays)

with newer technology (CT), its speed and ease to obtain, concerns about malpractice and other factors, and for reasons that are surely multifactorial [18]. With hospital crowding, time pressures, the need to make decisions based on limited information, and a mandate to never miss life-threatening disease, the ED is fraught with the potential for error. This generally leads to a bias toward testing, including advanced imaging. It is important to appreciate that the increases in CT imaging in the ED have occurred in the context of a significant shift in the USA in the setting in which acute care is provided. The ED now accounts for nearly a third of the 354 million annual acute care visits in the USA, practically all acute care provided after hours and on weekends, more acute care for the uninsured than in all other settings combined, and nearly 50% of hospital admissions in the USA [26, 27]. To some degree, this shift in location of care explains some of the increases in volume of imaging performed in this setting, though early studies demonstrated that increases in CT utilization outpaced increases in ED visits [28]. Studies demonstrate that increased CT utilization far exceeds the amounts expected by replacement of technology [29] and that changes in tort law have had mixed findings as relates to CT utilization [30, 31].

Patterns of Repeat/Multiple CT Imaging in the ED

Though head and abdominopelvic CTs are the most commonly performed studies in the ED, it is known that certain indications for CT (e.g., kidney stones, suspected pulmonary embolus) predominate among patients who are repeatedly or multiply imaged [25, 32, 33]. It has been observed that significant proportions of CTs performed are unnecessary and/ or could be replaced with other imaging modalities that do not impart radiation [34–37]. Studies identifying patients who are heavily imaged at a given point in time find that these patients tend to have multiple imaging generally and patients heavily imaged in the ED are also highly imaged in settings outside the ED [32, 33, 38].

Radiation and Its Effects

Radiation is the passage of an electromagnetic wave through space. At one end of the electromagnetic spectrum, lowerfrequency and lower-energy waves, including radio waves, microwaves, infrared, visible and ultraviolet light, as well as ultrasound, comprise *non-ionizing radiation*. At the other end of the spectrum, the higher-energy, higher-frequency waves, including X-rays and gamma rays, are considered *ionizing radiation*. X-rays are produced when electrons are emitted from electron clouds as a result of electron excitation. Gamma rays are emitted from unstable nuclei as part of radioactive decay. Radiation is considered ionizing if it is of high enough energy to remove electrons from an atom. This is the basis for ionizing radiation causing cellular injury at the atomic and molecular level.

A displaced electron can cause direct injury if it hits and damages a strand of DNA or can indirectly damage DNA if the electron reacts with water, causing a hydroxyl radical that then interacts with DNA. When just a single strand of DNA is damaged, the cell is usually able to repair this, but when both strands are damaged, an abnormal reconnection of strands can occur, which is believed to account for the negative effects of radiation in humans. This may include a rejoining of strands incorrectly, leading to cell death or rejoining as a symmetrical translocation. This can result in oncogene expression during division and subsequent development of a malignancy or abnormal division in the gonads, potentially leading to hereditary disorders [39].

Deterministic Versus Stochastic Effects of Ionizing Radiation

At high doses, high-energy radiation causes direct cellular injury, resulting in what are called *deterministic effects*. These effects are dose related, occurring at threshold levels of radiation to cause sufficient cellular death within a tissue that then results in functional impairment of that organ or tissue (Fig. 10.3). Deterministic effects are discrete and specific, occurring within specific time frames following exposure. These effects are typically due to single large overdoses of radiation. Some examples include:

- · Skin erythema, necrosis, and sloughing
- Cataract formation
- Sterility
- Radiation illness (GI tract, bone marrow, CNS)
- IUGR, teratogenesis, and fetal death

By comparison, so-called stochastic effects of radiation are due to DNA mutations whose effects become apparent after cell division. By definition, these are random, not guaranteed to occur, and they develop in an unclear time frame. Stochastic effect occurrence is thought to follow a linear, nothreshold (LNT) exposure where cumulative low doses result in increasing risk, not requiring some threshold level in order to cause these effects. **Fig. 10.3** Examples of deterministic effects of ionizing radiation. (From ICRP, 1991. 1990 Recommendations of the International Commission on Radiological Protection. ICRP Publication 60. Ann. ICRP 21 (1–3), with permission)



The Link Between Low-Dose Ionizing Radiation and Cancer

Without the benefit of direct observation to quantify amounts, considerable debate has existed about the nature of radiation exposure and the relationship to cancer. Ouestions about the relationship between radiation and cancer include whether cancer due to radiation results only from discrete exposures to some threshold amount, or whether the risk of cancer due to low levels of radiation increases in a linear (or other) fashion. Studies exploring this relationship have been based largely on data from observed vs. expected solid and liquid cancer rates among those in the blast zones of Hiroshima and Nagasaki, approximately 2000-3000 yards from ground zero receiving radiation in the 5-100 mSv range. Among 93,000 survivors followed over 55 years, 31,650 received a dose of 5-100 mSv. 44% of leukemia and 8% of solid tumor cases were attributable to radiation [40, 41]. Studies among survivors of the Chernobyl nuclear accident have also established the role of radiation as a precursor to malignancy. Other studies report increased all-cause mortality (primarily due to dose-related increases of cancer mortality) of nuclear workers related to cumulative low-dose exposure [42] and both projected cancers [43] and effects on cognition among infants exposed to low-dose radiation [44, 45]. In 2005, the US National Research Council's Seventh Biologic Effects of Ionizing Radiation Conference (BEIR VII) adopted the most widely used risk model for the effects of low levels of ionizing radiation [46]. The BEIR VII model, accepted by the International Commission on Radiological Protection and the UN Scientific Committee on the Effects of Atomic Radiation, holds that the risk of cancer from cumulative lowlevel radiation proceeds in a linear, no-threshold fashion so that low levels of radiation are cumulative and do not require a threshold level to increase the risk of cancer (Fig. 10.4)



Fig. 10.4 Possible dose response curves for cancer induction at low doses. (a) Supralinear, (b) LNT (linear non-threshold), (c) threshold, (d) hermetic. Summing up the epidemiological data and biological findings, we are still unsure about the shape of the dose response curve in the low-dose range. While several dose response curves are possible as illustrated in the figure, the LNT model is usually adopted as a best estimate for low-dose risk. (From Anzai et al. [47], with permission Editorial Secretariat for JCBN)

[47]. Further, this risk model states that a 10 mSv exposure increases the risk of cancer by approximately 1/1000. In the USA, where the lifetime risk of cancer is approximately 42%, this means very crudely that a cumulative 10 mSv exposure increases this risk to 42.1%. Approximately half of cancers are fatal, and so by this risk model, a 10 mSv exposure translates to a risk of fatal cancer of approximately 1/2000 (Fig. 10.5) [46]. This model remains highly controversial, with major bodies, including the French Academy of Sciences, the American Nuclear Society, and the National Academy of Medicine suggesting that this model overesti-



Fig. 10.5 In a lifetime, approximately 42 (solid circles) of 100 people will be diagnosed with cancer from causes unrelated to radiation. The calculation in this report suggests approximately 1 cancer in 100 people could result from a single exposure 100 mSv of low-LET radiation [46]

mates risk. It is important to appreciate that though this remains controversial, it is nonetheless the most widely accepted risk model. It is important to recognize that this model is unadjusted for gender and age at exposure. The American Association of Physicists in Medicine considers a cumulative dose in excess of 100 mSv to be of concern. The quantity of radiation that is considered to be "low dose" is generally in the range of 5–100 mSv.

Who Is Most at Risk?

It is known that age at exposure and gender are modifiers of risk. Women are scanned more frequently and are more sensitive to the effects of radiation, primarily due to increased radiosensitivity of female gonadal tissues [25]. Children have both a longer lag time to develop mutations or for mutations to result in malignancies, as well as increased radiosensitivity that declines with age [24]. However, recent models suggest that this decrease continues until middle age but that then cancer risks may then increase in a U-shaped distribution [15]. An additional risk factor is high cumulative radiation exposure due to multiple imaging. Certain conditions are known to be at increased risk for multiple imaging [32, 33, 48–53] including but not limited to inflammatory bowel disease (IBD) [51–55], kidney stones [20, 56, 57], and

shunted hydrocephalus. This leads, in some patients, to very high cumulative doses and increases in lifetime attributable risk (LAR) of cancer. Patients who already have cancer may be at increased risk to develop second malignancies due to radiation. This is documented for radiotherapy, particularly for the lung, esophagus, luminal cancers [58], sarcoma, and breast cancer [59–61]. Cumulative low-dose radiation could represent another potential source of risk.

Measures of Radiation

There are many different units of radiation in use, and these can be difficult to remember for those who do not use them regularly. Measures of radiation relate to different aspects of radiation: exposure, absorption, biological effects, and for comparing effects and values of different exposures.

- *Exposure* to radiation can be measured in roentgens [R]. Exposure is the strength of a radiation field at some point in the air.
- Absorption is entry of radiation into body tissues and is measured in rads where 1 rad = 100 ergs/g or the International System of Units (SI) version and grays (Gy). A gray is defined as the absorption of 1 J (joule) of energy by 1 kg (kilogram) of matter. One gray = 100 rad.



Fig. 10.6 Units and naming conventions for ionizing radiation. (From https://commons.wikimedia.org/wiki/File:SI_Radiation_dose_units.png with permission. Accessed: 8 Jul 2020 Courtesy of Doug Sim CC By-SA 3.0)

Absorbed dose is therefore expressed as a concentration, not an absolute amount. Absorption depends on the strength of the radiation, the distance from the source, and the duration of the exposure.

- The *biological effects* of absorbed radiation are measured in rems or sieverts (Sv), which is the SI unit.
- Important but sometimes confusing concept that is widely used and that can be somewhat controversial is that of *effective dose* (Fig. 10.6). This relates absorbed dose and biological effects. Effective dose is also used in comparing cancer risks and is not a dose but is rather a concept to reflect the risk of cancer from an exposure reflected over age and gender. It is expressed in millisieverts. Effective dose is determined by taking the sum of organ or tissue doses, multiplying these by the respective tissue weighting factors, based on an assumed uniform, whole body exposure. Because it is an average, unadjusted for age and gender, and mathematically derived using a standard body, it is not a true reflection of risk for an individual (Table 10.1).

Dose Estimates

Common units in estimating radiation dose include CT dose index (CTDI) and dose length product (DLP). Most current CT scanners generate these data based on the information provided in determining the protocol used, and these metrics can be reported with interpretations or in other lags.

Reporting of CTDI and DLP may soon be required for all scans. CTDI can be thought of as the amount of radiation dose imparted in a single axial CT "slice" through the body with associated scatter and is measured by radiation detectors in a standardized acrylic "phantom" designed to detect doses to specific organs. Different variants of this measure exist, but the volume CTDI is most commonly used, measuring the amount of radiation delivered to the scan volume of a standardized phantom. DLP is the slice thickness multiplied by the number of slices acquired or length of the body scanned. When considering doses of radiation provided by a scanner, it should be remembered that these are not direct measurements for the patient on the gantry, but rather are estimates based on the protocol used as determined by detection in phantoms. In addition to age and gender, weight or body habitus also impacts the absorbed dose.

Efforts to Reduce Radiation and Optimize Imaging

Technological Improvements

At its most basic, CT is an X-ray tube that rotates around a patient. The X-rays passing through the patient are attenuated differently by different body tissues resulting in the detection of a pattern of photons by the detector opposite the tube. As a patient passes through the scanner, it acquires numerous "pic-

		ICRP 60		IRCP 103			Sources for comparison		
Category	Examination	Mean	Min–Max	SD	Mean	Min–Max	SD	Mettler et al. [62]	DDM2 [63]
Computed tomography	Abdomen	8.1	5.1-11.7	2.0	6.8	5.6-8	1.2	8.0	11.3
	Chest	6.7	4.4-11.8	2.1	7.0	4.6-10.1	1.7	7.0	6.6
	Head	1.8	1.4-2.6	0.4	1.7	0.9–2.6	0.5	2.0	1.9
	Neck	3.2	1.8-6.0	1.3	3.0	1.7-5.8	1.9	3.0	2.5
	Pelvis	8.3	4.0-11.9	2.4	7.4	5.7–9.9	2.2	6.0	7.3
	Spine	10.3	4.0–16.7	5.3	7.0	1-12	0.0	6.0	7.7
	Trunk	12.2	6.7–15.8	3.3	12.3	10–16	2.0	-	14.8
Interventional cardiology	PCTA	19.5	7.4-48.6	15.1	7.2	-	-	15.0	15.2
Plain radiography	Abdomen	0.92	0.21-2.1	0.6	0.5	0.14-0.75	0.25	0.7	0.9
	Cervical spine	0.08	0.02-0.18	0.06	0.05	0.01 - 0.11	0.05	0.20	0.19
	Chest/thorax	0.07	0.01 - 0.14	0.04	0.05	0.01 - 0.07	0.02	0.02	0.10
	Lumbar spine	1.2	0.2-1.9	0.6	0.80	0.2-1.5	0.70	1.5	1.2
	Mammography	0.33	0.26-0.46	0.11	0.64	-	-	0.40	0.27
	Pelvis and hip	0.90	0.45-1.82	0.47	0.37	0.09-0.66	0.24	0.60	0.71
	Thoracic spine	0.60	0.23-1.22	0.43	0.50	0.1-1.2	0.40	1.00	0.64
Fluoroscopy	Ba enema	5.8	3.0-8.25	2.4	2.9	2.2-3.5	0.90	8.0	8.5
	Ba follow	3.5	1.2-7.7	3.7	1.3	1.2–1.3	0.10	5.0	7.3
	Ba meal	3.6	1.5-4.93	1.5	4.5	-		6.0	6.2
	Cardiac angiography	9.3	3.3-22.3	6.4	3.1	-		7.0	7.7
	Urogram	3.5	2.3-6.5	2.0	2.1	-		3.0	2.9

Table 10.1 Mean effective dose (mSv) for the 20 selected examinations^a

From Vilar-Palop et al. [64], with permission CC by 3.0

Note: Min, max, and standard deviation are shown only when more than one value was found

DDM2 Dose Datamed 2, PCTA percutaneous transluminal coronary angioplasty, Ba barium

^aAccording to the two different sets of the International Commission on Radiological Protection values used (ICRP 60 from 1990 and ICRP1 03 from 2007) and the values of two sources for comparison

tures" in a 360-degree helical fashion that are then mathematically computed and joined to create an image that can be reconstructed in three dimensions. A number of technological improvements can greatly reduce dose per study:

- Standardizing the doses and protocols of CT studies so that for a given patient and a given study, the same dose is delivered. One study of four hospitals in the SF area found that for the same CT study in the same patient, doses varied by up to 13-fold. This sort of variability had significant implications as to projected cancer risk [65] and is unnecessary and wasteful.
- Optimizing aspects of data acquisition (e.g., speed with which the table passes through the scanner, length of body scanned, pitch at which patient passes through the scanner, minimizing amount of overlapped areas occurring due to the helical nature of the scan) may help reduce exposure.
- New detector technology, increasing detector number, and dual source imaging also offer the potential for lower doses [66].
- Accepting more noise in images where high resolution is unnecessary, such as for detecting ureteral stones, can decrease the radiation associated with this CT [67].

- Minimizing multiphase scanning, where patients are scanned, for example, first without, then again with, a contrast agent.
- Modulating tube current, where the amperage from the X-ray tube of a CT scanner is modified as it spins around the patient on the gantry such that a sufficient higher amperage is used in the plane in which the patient is wider and a lower mA setting is less used in the plane in which the patient is thinner.
- Use of shields, such as breast shields, may reduce dose to sensitive tissues. However, these also have limitations and can introduce noise and artifacts.
- Other technical features such as iterative reconstruction can provide high-quality images by eliminating sources of noise, for example, in the reconstruction of the data into images.

Though technological solutions hold promise to minimize the dose per scan, the biggest payoff in terms of radiation and certainly with respect to costs associated with imaging may come at the point of order entry and optimizing ordering to scan those who need it and avoid scanning those who do not. Changing provider behavior is difficult to do, however, as a number of different efforts have demonstrated.

Provider and Patient Awareness

Numerous studies have demonstrated lack of physician awareness of the risks of imaging, disbelief in risks of cancer related to radiation from imaging, and poor performance in estimating risks and equivalent doses of radiation between imaging modalities. This lack of knowledge crosses specialties, in some cases even including radiologists [68-73]. While these are mostly older studies that predate the current focus on this area, the few studies specific to EPs confirm these findings [68, 74, 75]. That said, it is not entirely clear what the right knowledge is: does it matter whether the EP knows the number of chest X-rays that are equivalent to an abdominal CT, or is it more important to know the increase in lifetime attributable risk imparted by an abdominal CT? What is the level of knowledge required for good decisionmaking? Raising awareness, though helpful, is likely insufficient on its own to make a difference. In one study, training house staff on radiation risks failed to change ordering behavior, though it made them more comfortable with discussing risks with patients [76].

Approaches to raising awareness have included a number of efforts providing information cards for physicians that outline radiation risk for patients, though studies of the effectiveness of these have not been published in the medical literature. One successful program in raising awareness is the Image Gently campaign, begun in 2007, and directed at improving the safety and effectiveness of the imaging of children. This includes "raising patient and provider awareness, providing education and advocacy on selection of appropriate imaging studies, and minimizing radiation dose to levels as low as reasonably achievable (ALARA) when imaging children" [77]. This successful campaign has subsequently led to the Image Wisely campaign directed at achieving similar goals for adults [78].

The few studies that have looked specifically at ED patient knowledge about radiation and their preferences found that patients generally prefer imaging and definitive diagnoses without concern for radiation or its risks [10, 11]. Studies among pediatric populations found that discussions of radiation risk with parents improved their understanding without causing them to decline necessary imaging studies [79]. In a survey of radiology department chairs at academic medical centers, two-thirds reported having guidelines at their institutions related to informed consent for non-emergent CT studies. And though informed decision-making related to imaging is recommended as a best practice [80, 81], only 15% included discussions about possible radiation risk of CT with their patients [82].

Clinical Decision Support

Appropriateness Criteria

A number of bodies have developed appropriateness criteria for imaging [83, 84], perhaps most notable of these is the American College of Radiology (ACR) Appropriateness Criteria, which consist of consensus-derived rankings of imaging studies and modalities for specific clinical indications for imaging. Particularly when radiologists are not available for consultation, rankings of appropriateness may help guide imaging selection and potentially curb overutilization. Providers board-certified in emergency medicine may find these criteria less useful in selecting appropriate imaging in the ED as this is part of their training and because daily practice and local resource availability inform selection of appropriate imaging modalities. In addition, rankings in the ACR Appropriateness Criteria are not impacted by the amount of radiation a study imparts, which is listed in a separate column called "relative radiation level."

Clinical Decision Rules

Clinical decision rules (CDRs) have been developed to help guide clinical practice in an evidence-based manner. Though this term is sometimes used casually, scientifically sound clinical decision rules require rigorous derivation and validation methods that are time-consuming and costly, requiring expertise to ensure their ultimate appropriate use. Moreover, it is unclear whether CDRs truly outperform clinical gestalt [85, 86]. CDRs relating to imaging have been developed to help determine whether patients require head CT imaging following minor traumatic brain injury, whether patients suspected for possible pulmonary embolus require chest CT, and whether patients with renal colic should undergo CT imaging [37, 87]. Even when a CDR is demonstrated to be effective, valid, and reliable, it is only useful if it is actually used. A number of studies describe how various imaging decision rules are underutilized [88-90]. One study estimated that use of a decision rule could prevent up to onethird of pulmonary embolus CTs [34].

Computerized Interventions

A common way to promote use of CDRs and appropriateness criteria is to embed them in computerized order entry systems. A limited number of studies have explored the

effectiveness of computerized decision support in the ED. There have been limited or no studies evaluating use of computerized decision support in providing information on the dose associated with commonly ordered CT studies, patients' individual CT study counts, associated cumulative radiation exposure, or lifetime attributable risks of cancer. One study described computer-assisted identification of patients who had accumulated certain threshold numbers of lifetime CTs, which then required a peer-to-peer conversation with a radiologist to proceed [91]. Though it has been proposed that patients might carry a card with them that specifies the amount of cumulative radiation they have had, it is unclear what providers should do with this information. One survey of EPs confirmed that though providers were interested in all forms of decision support and information described above, they were inadequately familiar with information on radiation dose to make use of such information clinically.

Shared Decision-Making with Patients

Emergency physicians want computerized tools to help guide decision-making and would like information that provides them with ways to discuss risks with their patients [75]. Data suggest that despite being recommended as best practice, discussions with patients regarding risks rarely occur [92]. A main limitation of shared decision-making as relates to CT imaging in the ED is that the decision to image is infrequently preference sensitive or at equipoise with some other diagnostic option. Conveyance of complex information is challenging enough in patients with adequate literacy and numeracy, let alone among those with lesser skills.

Quality Metrics and Regulatory Efforts

In addition to physician-initiated efforts to optimize imaging, a number of organizations and regulatory bodies have taken an interest in incentivizing improvements. Preauthorization is one approach taken by payers that has been successful in reducing imaging. However, it is frustrating for physicians to seek approval to care for their patients and would be highly impractical for use in the ED. Quality measures related to imaging have been advanced by various bodies, including proposals to report and track patientspecific dose information. Existing or proposed legislation in some states involves tracking CTDI and DLP. In addition, tracking and documentation of CTDI and DLP in each patient's record are required by the TJC for accreditation. Hospitals must compile and analyze data on patient CT radiation doses and compare these with external benchmarks when available.

The Centers for Medicare and Medicaid Services (CMS) have adopted imaging-related measures for the ED included as part of the Quality Payment Program. Other programs such as the Choosing Wisely campaign that focus on limiting avoidable imaging have also been championed in emergency medicine. One of these measures, directed at utilization of head CT in the ED for atraumatic headache, generated a fair amount of debate in the EM community. This measure, OP-15, was controversial because its derivation applied data obtained in a younger population to an older Medicare population, its exclusion criteria were felt to be insufficient, and it failed to include indications that were included in the American College of Emergency Physicians' clinical policies related to atraumatic headache and thus was felt to be invalid for public reporting. The Choosing Wisely initiative includes five recommendations directed at minimizing diagnostic radiation exposure: (1) avoiding head CT scans in patients with minor head injury who are at low risk based on validated decision rules; (2) avoiding head CT in asymptomatic adult patients with syncope, insignificant trauma, and a normal neurological evaluation; (3) avoiding CT pulmonary angiography for patients with low pretest probability of pulmonary embolism and either a negative Pulmonary Embolism Rule-Out Criteria (PERC) score or a negative D-dimer; (4) avoiding lumbar spine imaging for adults with non-traumatic back pain unless the patient has severe or progressive neurologic deficits or is suspected of having a serious underlying condition (such as vertebral infection, cauda equina syndrome, or cancer with bony metastasis); and (5) avoiding abdominopelvic CT in young otherwise healthy patients with known histories of kidney stones or ureterolithiasis, presenting with symptoms consistent with uncomplicated renal colic [93].

Summary

It is hard to imagine the practice of emergency medicine without the use of CT imaging. But it may be very important for the benefit of patients, to imagine practice with more judicious use of this tool, particularly for those at increased risk of cancer. Although the nature of the relationship of lowdose radiation and cancer remains controversial, it is accepted that radiation is a carcinogen and that high cumulative doses increase the risk of cancer. Awareness of the risks of CT imaging in young and female patients, particularly for those at repeat and multiple imaging, is a start toward appropriate use. A number of technological improvements are promising to reduce the dose per study. Decision support tools may be useful in helping risk-stratify patients to help in decisionmaking; however, more study is needed to evaluate their effectiveness. Quality measures and regulatory statutes related to tracking of cumulative doses may also help drive improvement. Much work remains in order to make clinical use of this information. Finally, providing emergency physicians with options to CT imaging for high-risk patients, including alternate imaging modalities, is essential to help reduce utilization.

References

- Webb EM, Nguyen A, Wang ZJ, Stengel JW, Westphalen AC, Coakley FV. The negative appendectomy rate: who benefits from preoperative CT? AJR Am J Roentgenol. 2011;197(4):861–6.
- Rao PM, Rhea JT, Novelline RA, Mostafavi AA, McCabe CJ. Effect of computed tomography of the appendix on treatment of patients and use of hospital resources. N Engl J Med. 1998;338(3):141–6.
- Kim K, Lee CC, Song KJ, Kim W, Suh G, Singer AJ. The impact of helical computed tomography on the negative appendectomy rate: a multi-center comparison. J Emerg Med. 2008;34(1):3–6.
- Coursey CA, Nelson RC, Patel MB, Cochran C, Dodd LG, Delong DM. Making the diagnosis of acute appendicitis: do more preoperative CT scans mean fewer negative appendectomies? A 10-year study. Radiology. 2010;254(2):460–8.
- Balthazar EJ, Rofsky NM, Zucker R. Appendicitis: the impact of computed tomography imaging on negative appendectomy and perforation rates. Am J Gastroenterol. 1998;93(5):768–71.
- Kocher KE, Meurer WJ, Fazel R, Scott PA, Krumholz HM, Nallamothu BK. National trends in use of computed tomography in the emergency department. Ann Emerg Med. 2011;58(5):452–62. e3.
- Hoffmann U, Truong QA, Fleg JL, Goehler A, Gazelle S, Wiviott S. Design of the Rule Out Myocardial Ischemia/Infarction Using Computer Assisted Tomography: a multicenter randomized comparative effectiveness trial of cardiac computed tomography versus alternative triage strategies in patients with acute chest pain in the emergency department. Am Heart J. 2012;163(3):330–8. 338.e1.
- Abujudeh HH, Kaewlai R, McMahon PM, Binder W, Novelline RA, Gazelle GS. Abdominopelvic CT increases diagnostic certainty and guides management decisions: a prospective investigation of 584 patients in a large academic medical center. AJR Am J Roentgenol. 2011;196(2):238–43.
- Baumann BM, Chen EH, Mills AM, Glaspey L, Thompson NM, Jones MK. Patient perceptions of computed tomographic imaging and their understanding of radiation risk and exposure. Ann Emerg Med. 2011;58(1):1–7.e2.
- Takakuwa KM, Estepa AT, Shofer FS. Knowledge and attitudes of emergency department patients regarding radiation risk of CT: effects of age, sex, race, education, insurance, body mass index, pain, and seriousness of illness. AJR Am J Roentgenol. 2010;195(5):1151–8.
- Lee DW, Foster DA. The association between hospital outcomes and diagnostic imaging: early findings. J Am Coll Radiol. 2009;6(11):780–5.
- Lichtenberg FR. The quality of medical care, behavioral risk factors, and longevity growth. Int J Health Care Finance Econ. 2011;11(1):1–34.
- Wears RL. The hunting of the snark, 2011 (editorial). Ann Emerg Med. 2011;58(5):465–7.
- National Council on Radiation Protection & Measurements. Ionizing radiation exposure of the population of the United States, Report No. 160. Bethesda: National Council on Radiation Protection & Measurements; 2009.
- Redberg RF, Smith-Bindman R. We are giving ourselves cancer. New York Times. 30 Jan 2014.

- Levin DC, Rao VM, Parker L. Physician orders contribute to high-tech imaging slowdown. Health Aff (Millwood). 2010;29(1):189–95.
- Larson DB, Johnson LW, Schnell BM, Salisbury SR, Forman HP. National trends in CT use in the emergency department: 1995– 2007. Radiology. 2011;258(1):164–73.
- Levin DC, Rao VM, Parker L. The recent downturn in utilization of CT: the start of a new trend? J Am Coll Radiol. 2012;9(11):795–8.
- Fazel R, Krumholz HM, Wang Y, Ross JS, Chen J, Ting HH. Exposure to low-dose ionizing radiation from medical imaging procedures. N Engl J Med. 2009;361(9):849–57.
- Broder J, Bowen J, Lohr J, Babcock A, Yoon J. Cumulative CT exposures in emergency department patients evaluated for suspected renal colic. J Emerg Med. 2007;33(2):161–8.
- Larson DB, Johnson LW, Schnell BM, Goske MJ, Salisbury SR, Forman HP. Rising use of CT in child visits to the emergency department in the United States, 1995–2008. Radiology. 2011;259(3):793–801.
- Smith-Bindman R, Miglioretti DL, Larson EB. Rising use of diagnostic medical imaging in a large integrated health system. Health Aff (Millwood). 2008;27(6):1491–502.
- 23. Schauer DA, Linton OW. NCRP Report No. 160, Ionizing radiation exposure of the population of the United States, medical exposure—are we doing less with more, and is there a role for health physicists? Health Phys. 2009;97(1):1–5.
- Brenner DJ, Hall EJ. Computed tomography—an increasing source of radiation exposure. N Engl J Med. 2007;357(22):2277–84.
- Berrington de Gonzalez A, Mahesh M, Kim KP, Bhargavan M, Lewis R, Mettler F. Projected cancer risks from computed tomographic scans performed in the United States in 2007. Arch Intern Med. 2009;169(22):2071–7.
- Schuur JD, Venkatesh AK. The growing role of emergency departments in hospital admissions. N Engl J Med. 2012;367(5):391–3.
- Pitts SR, Carrier ER, Rich EC, Kellermann AL. Where Americans get acute care: increasingly, it's not at their doctor's office. Health Aff (Millwood). 2010;29(9):1620–9.
- Broder J, Warshauer DM. Increasing utilization of computed tomography in the adult emergency department, 2000–2005. Emerg Radiol. 2006;13(1):25–30.
- 29. Griffey RT, Ledbetter S, Khorasani R. Changes in thoracolumbar computed tomography and radiography utilization among trauma patients after deployment of multidetector computed tomography in the emergency department. J Trauma. 2007;62(5):1153–6.
- Smith-Bindman R, McCulloch CE, Ding A, Quale C, Chu PW. Diagnostic imaging rates for head injury in the ED and states' medical malpractice tort reforms. Am J Emerg Med. 2010;29(6):656–64.
- Waxman DA, Ridgely MS, Heaton P. The effect of malpractice reform on emergency department care. N Engl J Med. 2015;372(2):192.
- Griffey RT, Sodickson A. Cumulative radiation exposure and cancer risk estimates in emergency department patients undergoing repeat or multiple CT. AJR Am J Roentgenol. 2009;192(4):887–92.
- Sodickson A, Baeyens PF, Andriole KP, Prevedello LM, Nawfel RD, Hanson R, Khorasani R. Recurrent CT, cumulative radiation exposure, and associated radiation-induced cancer risks from CT of adults. Radiology. 2009;251(1):175–84.
- 34. Venkatesh AK, Kline JA, Courtney DM, Camargo CA, Plewa MC, Nordenholz KE. Evaluation of pulmonary embolism in the emergency department and consistency with a national quality measure: quantifying the opportunity for improvement. Arch Intern Med. 2012;172(13):1028–32.
- Smith-Bindman R, Aubin C, Bailitz J, Bengiamin RN, Camargo CA Jr, Corbo J, et al. Ultrasonography versus computed tomography for suspected nephrolithiasis. N Engl J Med. 2014;371(12):1100–10.

- Griffey RT, Fowler KJ, Theilen A, Gutierrez A. Considerations in imaging among emergency department patients with inflammatory bowel disease. Ann Emerg Med. 2017;69(5):587–99.
- 37. Daniels B, Gross CP, Molinaro A, Singh D, Luty S, Jessey R, Moore CL. STONE PLUS: evaluation of emergency department patients with suspected renal colic, using a clinical prediction tool combined with point-of-care limited ultrasonography. Ann Emerg Med. 2016;67(4):439–48.
- Mettler FA Jr, Bhargavan M, Faulkner K, Gilley DB, Gray JE, Ibbott GS. Radiologic and nuclear medicine studies in the United States and worldwide: frequency, radiation dose, and comparison with other radiation sources—1950–2007. Radiology. 2009;253(2):520–31.
- Goodman TR. Ionizing radiation effects and their risk to humans. ImagingWisely.org. Image Wisely: Radiation safety in adult medical imaging. Nov 2010. https://www.imagewisely.org/-/media/ ImageWisely-Files/Imaging-Physicians/IW-Goodman-Ionizing-Radiation-Effects.pdf.
- Pierce DA, Preston DL. Radiation-related cancer risks at low doses among atomic bomb survivors. Radiat Res. 2000;154(2):178–86.
- Preston DL, Ron E, Tokuoka S, Funamoto S, Nishi N, Soda M, et al. Solid cancer incidence in atomic bomb survivors: 1958-1998. Radiat Res. 2007;168(1):1–64. https://doi.org/10.1667/RR0763.1. PMID: 17722996.
- 42. Cardis E, Vrijheid M, Blettner M, Gilbert E, Hakama M, Hill C, et al. The 15-Country Collaborative Study of Cancer Risk among Radiation Workers in the Nuclear Industry: estimates of radiationrelated cancer risks. Radiat Res. 2007;167(4):396–416.
- 43. Pearce MS, Salotti JA, Little MP, McHugh K, Lee C, Kim KP. Radiation exposure from CT scans in childhood and subsequent risk of leukaemia and brain tumours: a retrospective cohort study. Lancet. 2012;380(9840):499–505.
- 44. Hall P, Adami HO, Trichopoulos D, Pedersen NL, Lagiou P, Ekbom A. Effect of low doses of ionising radiation in infancy on cognitive function in adulthood: Swedish population based cohort study. BMJ. 2004;328(7430):19.
- 45. Raber J, Rola R, LeFevour A, Morhardt D, Curley J, Mizumatsu S. Radiation-induced cognitive impairments are associated with changes in indicators of hippocampal neurogenesis. Radiat Res. 2004;162(1):39–47.
- National Academy of Sciences/National Research Council. Health risks from exposure to low levels of ionizing radiation: BEIR VII, phase 2. Washington, DC: National Academies Press; 2006.
- Anzai K, Ban N, Ozawa T, Tokonami S. Fukushima Daiichi Nuclear Power Plant accident: facts, environmental contamination, possible biological effects, and countermeasures. J Clin Biochem Nutr. 2012;50(1):2–8.
- Desmond AN, O'Regan K, Curran C, McWilliams S, Fitzgerald T, Maher MM. Crohn's disease: factors associated with exposure to high levels of diagnostic radiation. Gut. 2008;57(11):1524–9.
- 49. Mc Laughlin PD, O'Connor OJ, O'Neill SB, Shanahan F, Maher MM. Minimization of radiation exposure due to computed tomography in inflammatory bowel disease. ISRN Gastroenterol. 2012;2012:790279.
- Newnham E, Gearry R, Gibson P. Factors associated with radiation exposure in patients with inflammatory bowel disease. Aliment Pharmacol Ther. 2010;31(4):534–5.
- 51. Newnham E, Hawkes E, Surender A, James SL, Gearry R, Gibson PR. Quantifying exposure to diagnostic medical radiation in patients with inflammatory bowel disease: are we contributing to malignancy? Aliment Pharmacol Ther. 2007;26(7):1019–24.
- Palmer L, Herfarth H, Porter CQ, Fordham LA, Sandler RS, Kappelman MD. Diagnostic ionizing radiation exposure in a population-based sample of children with inflammatory bowel diseases. Am J Gastroenterol. 2009;104(11):2816–23.

- 53. Peloquin JM, Pardi DS, Sandborn WJ, Fletcher JG, McCollough CH, Schueler BA, et al. Diagnostic ionizing radiation exposure in a population-based cohort of patients with inflammatory bowel disease. Am J Gastroenterol. 2008;103(8):2015–22.
- Chatu S, Poullis A, Holmes R, Greenhalgh R, Pollok RC. Temporal trends in imaging and associated radiation exposure in inflammatory bowel disease. Int J Clin Pract. 2013;67(10):1057–65.
- 55. Magro F, Coelho R, Guimarães LS, Silva M, Peixoto A, Lopes S, Macedo G. Ionizing radiation exposure is still increasing in Crohn's disease: who should be blamed? Scand J Gastroenterol. 2015;50(10):1214–25.
- Katz SI, Saluja S, Brink JA, Forman HP. Radiation dose associated with unenhanced CT for suspected renal colic: impact of repetitive studies. AJR Am J Roentgenol. 2006;186(4):1120–4.
- 57. Goldstone A, Bushnell A. Does diagnosis change as a result of repeat renal colic computed tomography scan in patients with a history of kidney stones? Am J Emerg Med. 2010;28(3):291–5.
- Hemminki K, Li X, Sundquist J, Sundquist K. Cancer risks in Crohn disease patients. Ann Oncol. 2009;20(3):574–80.
- Hall EJ, Wuu CS. Radiation-induced second cancers: the impact of 3D-CRT and IMRT. Int J Radiat Oncol Biol Phys. 2003;56(1):83–8.
- Maddams J, Parkin DM, Darby SC. The cancer burden in the United Kingdom in 2007 due to radiotherapy. Int J Cancer. 2011;129(12):2885–93.
- 61. Suit H, Goldberg S, Niemierko A, Ancukiewicz M, Hall E, Goitein M, et al. Secondary carcinogenesis in patients treated with radiation: a review of data on radiation-induced cancers in human, non-human primate, canine and rodent subjects. Radiat Res. 2007;167(1):12–42.
- Mettler FA Jr, Huda W, Yoshizumi TT, Mahesh M. Effective doses in radiology and diagnostic nuclear medicine: a catalog. Radiology. 2008;248(1):254–63.
- Health Protection Agency. Centre for Radiation, Chemical and Environmental Hazards. Luxembourg: Radiation Protection Unit; April 2008. http://www.sprmn.pt/pdf/RP154.pdf.
- Vilar-Palop J, Vilar J, Hernández-Aguado I, González-Álvarez I, Lumbreras B. Updated effective doses in radiology. J Radiol Prot. 2016;36(4):975–90.
- 65. Smith-Bindman R, Lipson J, Marcus R, Kim KP, Mahesh M, Gould R, et al. Radiation dose associated with common computed tomography examinations and the associated lifetime attributable risk of cancer. Arch Intern Med. 2009;169(22):2078–86.
- Henzler T, Fink C, Schoenberg SO, Schoepf UJ. Dual-energy CT: radiation dose aspects. AJR Am J Roentgenol. 2012;199(5 Suppl):S16–25.
- 67. Moore CL, Daniels B, Singh D, Luty S, Gunabushanam G, Ghita M, et al. Ureteral stones: implementation of a reduced-dose CT protocol in patients in the emergency department with moderate to high likelihood of calculi on the basis of STONE score. Radiology. 2016;280(3):743–51.
- Shiralkar S, Rennie A, Snow M, Galland RB, Lewis MH, Gower-Thomas K. Doctors' knowledge of radiation exposure: questionnaire study. BMJ. 2003;327(7411):371–2.
- McBride JF, Wardrop RM III, Paxton BE, Mandrekar J, Fletcher JG. Effect on examination ordering by physician attitude, common knowledge, and practice behavior regarding CT radiation exposure. Clin Imaging. 2012;36(5):455–61.e1.
- Jacob K, Vivian G, Steel JR. X-ray dose training: are we exposed to enough? Clin Radiol. 2004;59(10):928–34; discussion 926–7.
- Quinn AD, Taylor CG, Sabharwal T, Sikdar T. Radiation protection awareness in non-radiologists. Br J Radiol. 1997;70:102–6.
- 72. Rice HE, Frush DP, Harker MJ, Farmer D, Waldhausen JH. Peer assessment of pediatric surgeons for potential risks of radiation exposure from computed tomography scans. J Pediatr Surg. 2007;42(7):1157–64.

- Thomas KE, Parnell-Parmley JE, Haidar S, Moineddin R, Charkot E, BenDavid G, et al. Assessment of radiation dose awareness among pediatricians. Pediatr Radiol. 2006;36(8):823–32.
- Lee CI, Haims AH, Monico EP, Brink JA, Forman HP. Diagnostic CT scans: assessment of patient, physician, and radiologist awareness of radiation dose and possible risks. Radiology. 2004;231(2):393–8.
- Griffey RT, Jeffe DB, Bailey T. Emergency physicians' attitudes and preferences regarding computed tomography, radiation exposure, and imaging decision support. Acad Emerg Med. 2014;21(7):768–77.
- 76. Horowitz JM, Yaghmai V, Miller FH, Russell EJ. Will CT ordering practices change if we educate residents about the potential effects of radiation exposure? Experience at a large academic medical center. Acad Radiol. 2011;18(11):1447–52.
- Image Gently: the Alliance for radiation safety in pediatric imaging. https://www.imagegently.org/.
- Image Wisely: radiation safety in adult medical imaging. https:// www.imagewisely.org/.
- Larson DB, Rader SB, Forman HP, Fenton LZ. Informing parents about CT radiation exposure in children: it's OK to tell them. Am J Roentgenol. 2007;189(2):271–5.
- Brink JA, Goske MJ, Patti JA. Informed decision making trumps informed consent for medical imaging with ionizing radiation. Radiology. 2012;262(1):11–4.
- Goske MJ, Bulas D. Improving health literacy: informed decisionmaking rather than informed consent for CT scans in children. Pediatr Radiol. 2009;39(9):901–3.
- Lee CI, Flaster HV, Haims AH, Monico EP, Forman HP. Diagnostic CT scans: institutional informed consent guidelines and practices at academic medical centers. AJR Am J Roentgenol. 2006;187(2):282–7.
- American College of Radiology Appropriateness Criteria. http:// www.acr.org/Quality-Safety/Appropriateness-Criteria.
- World Health Organization. Effective choices for diagnostic imaging in clinical practice. Geneva: World Health Organization; 1990.

- 85. Hendriksen JM, Lucassen WA, Erkens PM, Stoffers HE, van Weert HC, Büller HR, et al. Ruling out pulmonary embolism in primary care: comparison of the diagnostic performance of "Gestalt" and the Wells rule. Ann Fam Med. 2016;14(3):227–34.
- Lucassen W, Geersing GJ, Erkens PM, Reitsma JB, Moons KG, Büller H, van Weert HC. Clinical decision rules for excluding pulmonary embolism: a meta-analysis. Ann Intern Med. 2011;155(7):448–60.
- 87. Moore CL, Bomann S, Daniels B, Luty S, Molinaro A, Singh D, Gross CP. Derivation and validation of a clinical prediction rule for uncomplicated ureteral stone – the STONE score: retrospective and prospective observational cohort studies. BMJ. 2014;348:g2191.
- Clement CM, Stiell IG, Lowe MA, Brehaut JC, Calder LA, Vaillancourt C, Perry JJ. Facilitators and barriers to application of the Canadian C-spine rule by emergency department triage nurses. Int Emerg Nurs. 2016;27:24–30.
- Brehaut JC, Stiell IG, Graham ID. Will a new clinical decision rule be widely used? The case of the Canadian C-spine rule. Acad Emerg Med. 2006;13(4):413–20.
- 90. Graham ID, Stiell IG, Laupacis A, McAuley L, Howell M, Clancy M, et al. Awareness and use of the Ottawa ankle and knee rules in 5 countries: can publication alone be enough to change practice? Ann Emerg Med. 2001;37(3):259–66.
- Birnbaum S. Radiation safety in the era of helical CT: a patientbased protection program currently in place in two community hospitals in New Hampshire. J Am Coll Radiol. 2008;5(6):714–718.e5.
- Vashi A, Rhodes KV. "Sign right here and you're good to go": a content analysis of audiotaped emergency department discharge instructions. Ann Emerg Med. 2010;57(4):315–22.e1.
- 93. Choosing Wisely. Ten things physicians and patients should question. The American College of Emergency Medicine. Last reviewed 2018. https://www.choosingwisely.org/societies/american-college-of-emergency-physicians/. Accessed 3 Jul 2020.

Cervical Cancer Screening

Natacha Phoolcharoen, Ellen S. Baker, and Mila Pontremoli Salcedo

Case Study

A 16-year-old girl received one dose of the human papillomavirus (HPV) quadrivalent vaccine when she was 14. The girl and her mother have come for vaccination counseling. They want to know if she can finish the HPV vaccination series and how many doses she will need. The nonavalent vaccine is available. Which vaccine should she receive, and how many doses are necessary to complete the series?

The only vaccine available now in the United States is a nonavalent vaccine, so she should receive another one dose of nonavalent vaccine. In the settings where a quadrivalent vaccine is still available, the clinician would recommend that she should receive another one dose of quadrivalent vaccine.

If providers do not know or do not have available the HPV vaccine product previously administered or are in settings transitioning to the nonavalent vaccine, any available HPV vaccine product may be used to continue or complete the series for females for protection against HPV 16 and 18.

Background

Cervical cancer is a leading cause of cancer death among women worldwide in spite of more than seven decades of effective prevention interventions. Cervical cancer is a preventable disease and its high mortality is unacceptable. In recognition of the morbidity and mortality associated with this preventable disease, in May 2018 the WHO Director General announced a global call for action toward the elimination of cervical cancer [1].

There were an estimated 569,847 new cases and 311,365 deaths from cervical cancer in 2018, and cervical cancer ranks as the fourth most commonly diagnosed cancer among women worldwide [2, 3]. The majority of cervical cancer cases occur in low- and middle-income countries (LMICs), where cervical cancer is often the most frequent cancer and the leading cause of cancer death among women. The global disparities in incidence and mortality of cervical cancer is due in part to the vast disparities in access to, and availability of, effective screening programs, including treatment of women who have precancerous lesions. Regions with the highest cervical cancer incidence rates are sub-Saharan Africa, Micronesia and Melanesia, Southeast Asia, Latin America, and the Caribbean [2].

Cervical cancer is a preventable disease. Primary prevention is available using one of the three available vaccines. If administered prior to sexual debut and exposure to HPV, generally in early adolescence, these vaccines will prevent 70–90% of cervical cancers. Furthermore, effective screening methods exist, including cytology and HPV testing. It takes many years for invasive cancer to develop following detection of a demonstrated persistent HPV infection and cervical dysplasia; thus there are many opportunities to intervene and treat a woman with abnormal screening test results. The main purpose of cervical cancer screening is to identify women with abnormal cervical lesions and treat precancerous lesions to prevent the progression to invasive cervical cancer.



N. Phoolcharoen (🖂)

Department of Gynecologic Oncology & Reproductive Medicine, The University of Texas MD Anderson Cancer Center, Houston, TX, USA

Department of Obstetrics and Gynecology, King Chulalongkorn Memorial Hospital, Faculty of Medicine, Chulalongkorn University, Bangkok, Thailand e-mail: nphoolcharoen@mdanderson.org

E. S. Baker

Department of Gynecologic Oncology & Reproductive Medicine, The University of Texas MD Anderson Cancer Center, Houston, TX, USA

M. P. Salcedo

Department of Gynecologic Oncology & Reproductive Medicine, The University of Texas MD Anderson Cancer Center, Houston, TX, USA

The Obstetrics and Gynecology Department, Federal University of Health Sciences of Porto Alegre (UFCSPA)/Santa Casa de Misericordia Hospital of Porto Alegre, Porto Alegre, Brazil

Human Papillomavirus (HPV)

HPV is the most common sexually transmitted infection. Extensive evidence links HPV and development of cervical cancer, and virtually all cervical cancer is caused by persistent infection with high-risk HPV [4]. The average lifetime probability of acquiring HPV is 85% and 91% among women and men who have had at least one sexual partner, respectively. More than 80% of women and men acquire HPV by age 45 years [5]. The initial infection generally occurs during adolescence or early adulthood. Approximately 80% of individuals with HPV will clear the infection spontaneously within 18–24 months of infection [6]. In women with persistent HPV infection, 3-5% will develop significant preinvasive disease, and <1% will develop cancer [7]. HPV infection is also associated with other malignancies, including oropharyngeal, anal, penile, vulvar, and vaginal cancers [8]. Persistent infection with a high-risk HPV (hrHPV) type is required for the development of cervical cancer.

HPV is a double-stranded DNA virus. There are over 120 HPV types, and approximately 40 types are known to infect the anogenital tract [9]. The HPV type distribution varies by population and geographic region [10].

HPV types are grouped by the potential of the virus to cause malignancy. Low-risk (non-oncogenic) HPV types do not integrate into the host cell genome. They cause genital warts, and low-grade cervical dysplasia. HPV types 6 and 11 are associated with 90% of genital warts. High-risk (oncogenic) HPV types are associated with high-grade cervical dysplasia and invasive cancer and may also be associated with low-grade cervical dysplasia. HPV types 16 and 18 account for 70% of cervical cancers and 60% of high-grade cervical dysplasia [11]. The DNA of high-risk HPV integrates into the host cell genome and causes neoplastic cellular changes. In most women, the HPV infection is not persistent, the virus is cleared naturally, and the dysplasia regresses. In a small percentage of women, the infection is persistent, and the transformed cells can replicate and progress to cancer after several years [11–13].

Cervical Intraepithelial Neoplasia (CIN)

Cervical intraepithelial neoplasia (CIN) is a precancerous lesion of the cervical epithelium. It is a histologic diagnosis made from a cervical biopsy specimen. In the Bethesda system for reporting cervical cytologic diagnoses (cervical cytology), findings are described with the term "squamous intraepithelial lesion (SIL)" and histologic findings described with the term "cervical intraepithelial neoplasia (CIN)" [14]. More recently, the 2019 American Society for Colposcopy and Cervical Pathology (ASCCP) Risk-Based Management Consensus Guidelines for abnormal cervical cancer screening tests and cancer precursors recommended histopathology reporting based on Lower Anogenital Squamous Terminology (LAST) group classification scheme. Per these guidelines, histologic high-grade squamous intraepithelial lesions (HSIL) should be classified with CIN 2 or CIN 3 qualifiers, i.e., HSIL (CIN 2) and HSIL (CIN 3) as outlined below [14, 15].

CIN is graded into three degrees of severity depending on the how much of the epithelial layer contains atypical cells. The terminology of CIN is shown in Table 11.1. CIN 1 frequently spontaneously regresses, often within 6–12 months [16]. CIN 2 remains reversible, with approximately 40% regressing spontaneously without treatment [17]. CIN 3, once called severe dysplasia and carcinoma in situ (CIS), is a precursor of invasive cancer. Approximately one-third of CIN 3 lesions may spontaneously regress. However, treatment is recommended for CIN 3 [15, 18].

Risk Factors

The lifetime risk of developing cervical cancer is 0.9% and 1.6% in high-income countries (HICs) and in LMICs, respectively [19]. The most frequent age at diagnosis in the United States is 35–44 years [20]. Women who are immunocompromised from any cause (e.g., genetic disease, HIV infection, immunosuppressive therapy) have a higher risk of developing a persistent HPV infection [21].

The primary risk factor for cervical cancer is persistent hrHPV infection. The cofactors associated with developing cervical cancer include immunosuppression, smoking, parity, increased number of sexual partners, and oral contraceptive use [22].

Prevention

Primary Prevention: HPV Vaccination HPV vaccines protect against acquiring HPV infection and the development of subsequent HPV-associated disease. Three different vaccines are currently commercially available worldwide, although not in all countries (Table 11.2) [23–26]. All three vaccines are prophylactic vaccines. Randomized clinical trials have demonstrated 93–98% efficacy in the prevention of cervical dysplasia associated with the HPV types contained in the vaccine [27–31]. Once exposed and infected with a specific hrHPV type, the vaccine is not effective in preventing disease against that type. The vaccine should be given prior to HPV exposure to be effective and protective. Currently, the only vaccine available in the United States is the nonavalent vaccine. The availability of vaccines varies in different countries and regions.

Bethesda classification	Cytology	LSIL	HSIL	
System	Histology	CIN 1	CIN 2	CIN 3
LAST Terminology	Cytology	LSIL	HSIL	
	Histology	LSIL (CIN 1)	HSIL (CIN 2)	HSIL (CIN 3)
Previous terminology		Mild dysplasia	Moderate dysplasia	Severe dysplasia/carcinoma in situ

Table 11.1 Terminology of cervical intraepithelial neoplasia

CIN Cervical intraepithelial neoplasia, LAST lower anogenital squamous terminology, LSIL low-grade squamous intraepithelial lesion, HSIL highgrade squamous intraepithelial lesion

Table 11.2	Comparison	of HPV	vaccines	[23-2]	6]	l
-------------------	------------	--------	----------	--------	----	---

	Bivalent	Quadrivalent	Nonavalent
Brand name	Cervarix TM	Gardasil™	Gardasil-9™
Protection	HPV 16/18	HPV 6/11/16/18	HPV 6/11/16/18/31/33/45/52/58
Cross-protection	HPV 31/33/45	HPV 31	No data
Adjuvant	ASO4	Aluminum hydroxyphosphate sulfate	Aluminum hydroxyphosphate sulfate

Recommendation The Advisory Committee on Immunization Practices (ACIP) recommends routine HPV vaccination for all girls and boys at 11–12 years, though children can receive vaccines as early as 9 years of age. Catch-up vaccination is recommended for females and males aged 13–26 years who have not received the vaccine or who have not completed the vaccination series. The US Food and Drug Administration (FDA) recently approved the use of Gardasil-9TM up to age 45 years [32], and the ACIP suggests shared clinical decision-making between patient and doctor when considering vaccination in this age group (26–45 years) [33].

Schedule and Doses

- Individuals starting the vaccination series before age 15: two doses (0, 6–12 months)
- Individuals starting the vaccination series at age 15 or older: three doses (0, 1–2 months, and 6 months).
- Immunocompromised individuals: three doses (0, 1–2 months, and 6 months), regardless of age

If the vaccination series is interrupted or not given as scheduled, the ACIP recommends resuming at any time without restarting the series, regardless of the gaps in the vaccination schedule.

Efficacy Six landmark phase III studies demonstrated that HPV vaccines are highly immunogenic and effective in preventing the main oncogenic types of HPV infections. The data were from three studies of the quadrivalent vaccine [27, 28, 34], two studies of the bivalent vaccine [29, 30], and one study of the nonavalent vaccine [31]. The vaccine induced antibodies in all vaccinated individuals, and antibody titers in vaccinated individuals were higher than those produced

by natural infection. Clinical trials of all three types of vaccines demonstrated an excellent humoral response with seroconversion rates near 95%.

The most recent evidence of HPV vaccine effectiveness is from large systemic reviews and meta-analyses, including data from 1702 articles with 60 million individuals and up to 8 years of follow-up. The data show the impact of HPV vaccines on HPV infections, CIN 2+, and anogenital warts. Among girls aged 13-19 years, the prevalence of HPV 16 and 18 decreased by 83%, and the incidence of anogenital warts decreased by 67%. In women aged 20-24 years, the prevalence of HPV 16 and 18 decreased by 66%, and the incidence of anogenital warts decreased by 54%. The incidence of CIN 2+ decreased significantly 5-9 years after vaccination by an estimated 50% among girls aged 15-19 years and 30% among women aged 20-24 years. In male participants, the incidence of anogenital warts decreased by 48% and 32% among boys aged 15-19 years and men aged 20–24 years, respectively [35].

The duration of immunity is not yet known, and so it is unclear if protection is life-long or whether a booster dose will be required. Because the vaccines do not provide protection against all cancer-associated HPV types, routine cervical screening is still recommended, including routine screening in vaccinated women.

Safety All HPV vaccines are made from virus-like particles (VLPs), which mimic the capsid of the virus but contain no genetic material [36]. These VLPs are not infectious and cannot cause an HPV infection. Several large clinical trials have documented all HPV vaccines to be safe, with no risks of serious adverse events. In addition, the World Health Organization (WHO) Global Advisory Committee on Vaccine Safety has stated that the benefit-risk profile is favorable [37].

Bivalent Vaccine Large placebo-controlled randomized trials data revealed the safety of the bivalent vaccine. There were no differences in serious adverse events between vaccine and placebo. In post-licensure data from the United States, there were 52 reports to the Vaccine Adverse Event Reporting System (VAERS) following administration of bivalent vaccine through September 2011, and 98% were considered nonserious [38].

Quadrivalent Vaccine The safety profile of the quadrivalent vaccine was evaluated in diverse populations of females from different resource settings in large licensing trials and post-licensure safety surveillance systems. Mild injection site reactions were the most commonly observed adverse events. The data showed that the vaccine is safe and well tolerated [27, 28].

There were 21,194 reports of adverse events following HPV immunization from 2006 to 2013, in which approximately 57 million doses of quadrivalent HPV vaccines were administered in the United States. Of reported adverse events, 92% were considered mild. Among serious events, headache, nausea, vomiting, fatigue, dizziness, syncope, and generalized weakness were the most frequently reported. There was no evidence of increased risk of Guillain-Barre syndrome compared with other vaccines [39].

Nonavalent Vaccine The nonavalent vaccine became available in 2014 and is currently the only vaccine used in the United States. There are less available post-licensure safety data of the nonavalent vaccine since it is relatively new. The overall safety profile appears similar to that of the quadrivalent vaccine with a slightly higher incidence of mild or moderate local reactions (pain, erythema, and swelling). The frequency of systemic adverse effects (e.g., headache, fever, nausea, dizziness) is similar to quadrivalent vaccines. Serious adverse effects occurred in less than 0.1% [40].

HPV Vaccination Challenges and Barriers In the United States, the uptake of HPV vaccination is relatively low, with only 50% of age-eligible children receiving the full series [41]. The acceptance of the vaccine in other high-income countries (HICs), e.g., Canada and Australia, is higher at approximately 70–80%, likely due to government-supported school-based programs [42–44]. The Global Alliance for Vaccination and Immunization (GAVI) has made the HPV vaccine available to low-income countries for US\$4 to \$5 per dose [45]. However, there are many remaining barriers limiting universal mass vaccination programs worldwide.

One of the barriers to HPV vaccination is cost, which is approximately \$150 per dose in the United States. Significant progress has been made to improve the affordability of the HPV vaccine in LMICs through financing mechanisms, including the GAVI and the Pan American Health Organization (PAHO) revolving fund [46]. However, many LMICs are still ineligible for financial support offered through the GAVI.

In many countries, a government supported school-based vaccination approach serves as an effective method to deliver the vaccine to adolescents. If girls do not attend school, the primary health centers and clinics can reach girls who missed vaccination [47].

General public knowledge of the importance of HPV vaccination is poor universally. There is widespread misinformation about the vaccine and the safety of vaccination, and some parents have been reluctant to have their children vaccinated. This is true even among well-educated populations in countries with high-quality healthcare systems [48].

Secondary Prevention: Cervical Cancer Screening

Cervical cancer screening to detect precancerous lesions, or CIN, enables treatment of pre-invasive lesions prior to development of invasive cancer. Screening tools include cervical cytology (Pap test), HPV testing, or a combination of the two tests. Systematic reviews and meta-analyses show that screening is associated with a significantly decreased incidence and mortality of invasive cervical cancer. Previous cluster-randomized studies from India revealed that a single screening test in a lifetime among women age 30-59 years decreased the risk of developing cervical cancer (relative risk 0.56, 95% confidence interval 0.42-0.75) and mortality from cervical cancer (relative risk 0.65, 95% confidence interval 0.47-0.90) [49, 50]. In many LMICs, Pap and HPV testing are not available. In these low-resource settings, visual inspection with acetic acid (VIA) is used to identify precancerous lesions of the cervix.

Cervical Cytology Dr. Papanicolaou demonstrated in 1928 that cancer and its precursors could be identified by examining an adequately prepared and stained cellular sample scraped from the uterine cervix [51]. This method of screening for cervical cancer precursors was adopted in many HICs in the 1940s. Although cervical cytology has low sensitivity, regular and widespread screening with cervical cytology has reduced the incidence of cervical cancer by 50-70% over the past 60-70 years [52]. Invasive cancer takes 10-20 years to develop from persistent HPV infection and resultant dysplasia, so screening can be an effective prevention strategy. Generally, women who develop invasive cervical cancer are women who have never been screened, or have not been screened regularly, or women who were screened, had abnormal cervical cytology, but did not receive follow-up treatment.

The conventional Pap test is performed by scraping cervical cells using a spatula and endocervical brush and smearing the cells on a glass slide for later examination by a cytologist/pathologist. In many countries in recent years, this method has been replaced with a liquid-based approach where the cervical sample is placed in a liquid medium initially, and the cells are later examined microscopically. This method allows for better preservation of the cervical cells and a higher-quality microscopic examination. The liquidbased collection method allows for cytology and HPV testing from the same sample.

HPV Testing HPV testing is now part of the routine screening algorithm in the United States. Studies have demonstrated that HPV testing has much higher sensitivity in detecting CIN 2+ than cervical cytology (96% vs. 53%) with slightly lower specificity (91% vs. 96%) [53–55]. HPV DNA testing has been recommended in both high- and lowresource countries [56, 57]. While cervical cytology is still widely used in LMICs, co-testing (cervical cytology and high-risk HPV testing) is now also recommended.

Several studies have examined the effectiveness of cervical cancer screening with primary HPV testing alone. In a large randomized study in Canada, 19,009 women were screened with HPV testing vs. liquid-based cytology. The study demonstrated that a woman with a negative HPV test vs. negative cytology had a significantly lower likelihood of CIN 3 and cancer at 48 months [58]. The high sensitivity of HPV testing leads to a much higher negative predictive value, implying that the screening interval can be safely lengthened if HPV testing is used [55]. **Guidelines for Screening** The ASCCP guidelines recommend screening for cervical cancer between the ages of 21 and 65 [59]. Cervical cancer screening should not be performed in women younger than 21 years of age, regardless of the age of onset of sexual activity. The screening guidelines, stratified by age, are summarized in Table 11.3

The US Preventive Services Task Force (USPSTF) recommends the primary high-risk HPV screening test as an alternative screening in the United States for women age 30–65 years. The USPSTF recommends screening with cervical cytology alone every 3 years, high-risk HPV testing alone every 5 years, or co-testing every 5 years in this group of women [60].

Colposcopy Women with abnormal cytology may be examined with colposcopy. The colposcope is a low-power binocular microscope with a powerful light source that is used for examination of the cervix. Acetic acid (3 to 5%) is applied to the cervix during this examination to enhance the visualization of dysplastic lesions [61]. A colposcopist can identify lesions and tissue patterns associated with cervical dysplasia and determine whether a biopsy is indicated.

Treatment of CIN Observation is preferred to treatment for women with CIN 1 [62]. In general, women with CIN 2/3 are treated because of the higher risk of progression to invasive cancer. Treatment of women with CIN 2 or CIN 3 to remove abnormal areas of the cervix may be through ablation (with cryotherapy or thermal ablation) or excision of the precancerous area. Success rates of these treatments are greater than 90% in properly selected patients [62, 63]. Excisional

 Table 11.3
 The American Cancer Society (ACS), American Society for Colposcopy and Cervical Pathology (ASCCP), and the American Society for Clinical Pathology (ASCP) summary of recommendations [59]

Population	Recommended screening methods	Notes
<21 years	No screening	Regardless of the age of onset sexual activity
21–29 years	Cervical cytology every 3 years	No HPV testing
30–65 years	Co-testing with cervical cytology and HPV every 5 years (preferred) or cervical cytology alone every 3 years	
>65 years	Screening is not recommended for women with a history of negative screening	Negative prior screening is defined as 3 consecutive negative PAP tests or 2 consecutive negative HPV tests, provided there is no history of high-grade dysplasia (CIN 2/3) or cancer in the past 20 years. Women 65 years of age or older who have not had previous screening should undergo cervical cytology and HPV testing
After hysterectomy with removal of the cervix	No screening	Screening is not recommended for women who have had a hysterectomy with removal of the cervix and do not have a history of CIN 2+
HPV vaccinated	Same as unvaccinated women	

CIN Cervical intraepithelial neoplasia

procedures include loop electrosurgical excision procedure (LEEP), cold knife conization (CKC), and CO2 laser conization. The choice of treatment modality depends on the availability of equipment and experience and expertise of the clinicians. In the United States, excisional treatment is preferred to ablative treatment for histologic HSIL (CIN 2 or CIN 3) [15]. Excision is also recommended for adenocarcinoma in situ (AIS) [62]. For specific treatment details, ASCCP guidelines should be consulted, as recommendations can change.

Cervical Cancer Prevention in Lowand Middle-Income Countries (LMICs)

The majority of cervical cancer cases occur in low- and middle- income countries (LMICs), where cervical cancer is a leading cause of cancer among women. Moreover, the death rate of cervical cancer in LMICs is 18 times higher than in HICs [64].

Although the screening and diagnosis algorithms described above are effective, they are expensive and require high-level infrastructure and well-trained personnel. In low-resource settings, personnel and medical specialists to provide prevention, screening, and treatment services are lacking, and shortages of infrastructure and pathology services are common. In several LMICs there are no gynecologic oncologists and no formal training in gynecologic oncologic. The International Gynecologic Cancer Society (IGCS) provides a Gynecologic Oncologic Global Curriculum and Mentorship Program, a comprehensive 2-year education and training program to train gynecologists in gynecologic oncologic in countries with no formal training program in this specialty [65].

In many LMICs there are no organized, effective cervical cancer screening programs, and few women ever receive cervical cancer screening. Furthermore, many women with abnormal screening results may be lost to follow-up and fail to receive diagnostic and therapeutic care for precancerous lesions [66].

One approach to screening in low-resource settings is the one-visit screen-and-treat strategy employing visual inspection with acetic acid (VIA) and immediate treatment with ablation if VIA screening is positive. VIA is performed by applying acetic acid to the cervix. Precancerous lesions will turn white with the application of acetic acid, and the identified lesion can be removed through ablation. Both VIA and ablation therapy (cryotherapy or thermal ablation) can be performed by nurses or trained community health workers [56, 57]. In some studies, screening with VIA was shown to decrease cervical cancer mortality by more than 30% in unscreened communities in LMICs [67, 68]. WHO also rec-

ommends HPV DNA testing as a primary cervical screening option in LMICs [56, 57].

Cervical Cancer Prevention in the Emergency Department

The emergency department (ED) can serve as a valuable public health resource. In the United States, the ED provides medical care to an estimated 60 million women, and approximately 20% of adults visit the ED for medical services each year [69, 70]. For underserved populations and people who lack access to regular medical services, the ED may be the only place some individuals receive healthcare services [71, 72]. In particular, underserved, low-income women are at increased risk of developing cervical cancer due to a lack of access to regular cervical cancer screening. Visits to the ED present an opportunity to provide, refer, or counsel women for cervical cancer prevention and screening.

A previous retrospective cross-sectional study of a stratified random sampling weighted to represent more than 100,000 visits to 4600 EDs showed the estimated median ED waiting time to be 30 minutes. The study also revealed that the median ED length of visit was 4.3 hours for admitted patients and 2.3 hours for discharged patients [73]. Some hospitals use these wait times as an opportunity for patient education in preventive health. Some centers use medical students, nursing staff, or health advocates to provide education in the waiting room. Moreover, education offered in the waiting room can be shared with families or caregivers and all visitors [74, 75].

HPV Vaccination

Physician recommendation of the HPV vaccine is a strong predictor of HPV vaccination initiation [76]. Recommendation of HPV vaccine in the ED is a potential channel to improve vaccination rates among populations who do not have access to routine medical care.

A cross-sectional study assessed attitudes of emergency medicine physicians toward recommending the HPV vaccine in an ED setting. The study surveyed the willingness of physicians to recommend the vaccine and identified barriers to vaccination in the ED settings. Of 100 physicians, 9% stated they would not recommend the vaccine to patients in the ED. Twenty-four percent were neutral, and 67% would recommend the vaccine. The study revealed that a lack of proper reimbursement for vaccination was a significant barrier to immunization in this setting [77]. Support and encouragement of emergency medicine physicians to inquire about, and recommend, HPV vaccination could be a useful step to improve primary prevention of cervical cancer.

Screening

Self-sample HPV testing has been proposed as an alternative method for cervical cancer screening in several settings. Women insert a small brush into the vagina to collect cervical samples without requiring a clinician-provided speculum exam. Many studies have shown the advantage of self-sample HPV testing in increasing screening rates and coverage [78, 79]. A previous study evaluated an acceptability of self-sample HPV testing among medically underserved women visiting the ED and found that 85% of participants were willing to use this method for cervical cancer screening [80]. Although this collection method is not yet approved by the US FDA, self-sampling HPV testing could be a potential strategy to improve cervical cancer screening among high-risk women who do not attend regular screenings.

A pilot study conducted at an urban ED found that referral combined with text messages resulted in a 43% increased uptake of cervical cancer screening among nonadherent groups [81]. The ED is a potential resource for improving HPV vaccination rates and cervical cancer screening rates. Patients can benefit from education about vaccination and cervical cancer screening and can also receive a referral for screening during a visit to the ED.

Conclusion

Cervical cancer remains an important public health problem in many countries and regions, including underserved areas of the United States. It is a disease that is preventable with HPV vaccination, regular cervical cancer screening, and follow-up care. However, access to care is a significant barrier for underserved women. Research suggests that the ED has great potential to serve as a resource for cervical cancer prevention.

References

- World Health Organization. Accelerating cervical cancer elimination 2018. Available from: https://www.who.int/reproductivehealth/ publications/screening-cervical-pre-cancer-lesions-women-with-hiv/en/
- Bray F, Ferlay J, Soerjomataram I, Siegel RL, Torre LA, Jemal A. Global cancer statistics 2018: GLOBOCAN estimates of incidence and mortality worldwide for 36 cancers in 185 countries. CA Cancer J Clin. 2018;68(6):394–424.
- Arbyn M, Weiderpass E, Bruni L, de Sanjose S, Saraiya M, Ferlay J, et al. Estimates of incidence and mortality of cervical cancer in 2018: a worldwide analysis. Lancet Glob Health. 2020;8(2):e191–203.
- Walboomers JM, Jacobs MV, Manos MM, Bosch FX, Kummer JA, Shah KV, et al. Human papillomavirus is a necessary cause of invasive cervical cancer worldwide. J Pathol. 1999;189(1):12–9.

- 5. Chesson HW, Dunne EF, Hariri S, Markowitz LE. The estimated lifetime probability of acquiring human papillomavirus in the United States. Sex Transm Dis. 2014;41(11):660–4.
- Moscicki AB. Management of adolescents who have abnormal cytology and histology. Obstet Gynecol Clin North Am. 2008;35(4):633–43; x.
- Schiffman M, Wentzensen N, Wacholder S, Kinney W, Gage JC, Castle PE. Human papillomavirus testing in the prevention of cervical cancer. J Natl Cancer Inst. 2011;103(5):368–83.
- Gillison ML, Chaturvedi AK, Lowy DR. HPV prophylactic vaccines and the potential prevention of noncervical cancers in both men and women. Cancer. 2008;113(10 Suppl):3036–46.
- 9. de Villiers EM, Fauquet C, Broker TR, Bernard HU, zur Hausen H. Classification of papillomaviruses. Virology. 2004;324(1):17–27.
- Moscicki AB, Hills N, Shiboski S, Powell K, Jay N, Hanson E, et al. Risks for incident human papillomavirus infection and low-grade squamous intraepithelial lesion development in young females. JAMA. 2001;285(23):2995–3002.
- Bosch X, Harper D. Prevention strategies of cervical cancer in the HPV vaccine era. Gynecol Oncol. 2006;103(1):21–4.
- Munoz N, Bosch FX, de Sanjose S, Herrero R, Castellsague X, Shah KV, et al. Epidemiologic classification of human papillomavirus types associated with cervical cancer. N Engl J Med. 2003;348(6):518–27.
- Munger K, Baldwin A, Edwards KM, Hayakawa H, Nguyen CL, Owens M, et al. Mechanisms of human papillomavirus-induced oncogenesis. J Virol. 2004;78(21):11451–60.
- Solomon D, Davey D, Kurman R, Moriarty A, O'Connor D, Prey M, et al. The 2001 Bethesda System: terminology for reporting results of cervical cytology. JAMA. 2002;287(16):2114–9.
- Perkins RB, Guido RS, Castle PE, Chelmow D, Einstein MH, Garcia F, et al. 2019 ASCCP risk-based management consensus guidelines for abnormal cervical cancer screening tests and cancer precursors. J Low Genit Tract Dis. 2020;24(2):102–31.
- Bansal N, Wright JD, Cohen CJ, Herzog TJ. Natural history of established low grade cervical intraepithelial (CIN 1) lesions. Anticancer Res. 2008;28(3B):1763–6.
- Castle PE, Schiffman M, Wheeler CM, Solomon D. Evidence for frequent regression of cervical intraepithelial neoplasia-grade 2. Obstet Gynecol. 2009;113(1):18–25.
- Ostor AG. Natural history of cervical intraepithelial neoplasia: a critical review. Int J Gynecol Pathol. 1993;12(2):186–92.
- Torre LA, Bray F, Siegel RL, Ferlay J, Lortet-Tieulent J, Jemal A. Global cancer statistics, 2012. CA Cancer J Clin. 2015;65(2):87–108.
- American Cancer Society. Key statistics for cervical cancer 2019 https://www.cancer.org/cancer/cervical-cancer/about/key-statistics. html. Last Medical Review: 16 Nov 2016. Last Revised: 30 Jul 2020. Accessed 25 Aug 2020.
- Ahdieh L, Klein RS, Burk R, Cu-Uvin S, Schuman P, Duerr A, et al. Prevalence, incidence, and type-specific persistence of human papillomavirus in human immunodeficiency virus (HIV)-positive and HIV-negative women. J Infect Dis. 2001;184(6):682–90.
- 22. Centers for Disease Control and Prevention. What are the risk factors for cervical cancer? 2018. Page last reviewed: 7 Aug 2019. https://www.cdc.gov/cancer/cervical/basic_info/risk_factors.htm. Accessed 25 Aug 2020.
- 23. Centers for Disease Control and Prevention (CDC). FDA licensure of bivalent human papillomavirus vaccine (HPV2, Cervarix) for use in females and updated HPV vaccination recommendations from the Advisory Committee on Immunization Practices (ACIP). MMWR Morb Mortal Wkly Rep. 2010;59(20):626–9.
- de Oliveira CM, Fregnani J, Villa LL. HPV vaccine: updates and highlights. Acta Cytol. 2019;63(2):159–68.
- Centers for Disease Control and Prevention. FDA licensure of quadrivalent human papillomavirus vaccine (HPV4, Gardasil)

for use in males and guidance from the Advisory Committee on Immunization Practices (ACIP). MMWR Morb Mortal Wkly Rep. 2010;59(20):630–2.

- 26. U.S. Food and Drug Administration. Vaccines, blood & biologics. vaccines. Gardasil 9. Content current as of 21 Aug 2020. https:// www.fda.gov/vaccines-blood-biologics/vaccines/gardasil-9. Accessed 25 Aug 2020.
- Garland SM, Hernandez-Avila M, Wheeler CM, Perez G, Harper DM, Leodolter S, et al. Quadrivalent vaccine against human papillomavirus to prevent anogenital diseases. N Engl J Med. 2007;356(19):1928–43.
- FUTURE II Study Group. Quadrivalent vaccine against human papillomavirus to prevent high-grade cervical lesions. N Engl J Med. 2007;356(19):1915–27.
- 29. Paavonen J, Naud P, Salmeron J, Wheeler CM, Chow SN, Apter D, et al. Efficacy of human papillomavirus (HPV)-16/18 AS04-adjuvanted vaccine against cervical infection and precancer caused by oncogenic HPV types (PATRICIA): final analysis of a double-blind, randomised study in young women. Lancet. 2009;374(9686):301–14.
- Hildesheim A, Wacholder S, Catteau G, Struyf F, Dubin G, Herrero R, et al. Efficacy of the HPV-16/18 vaccine: final according to protocol results from the blinded phase of the randomized Costa Rica HPV-16/18 vaccine trial. Vaccine. 2014;32(39):5087–97.
- Joura EA, Giuliano AR, Iversen OE, Bouchard C, Mao C, Mehlsen J, et al. A 9-valent HPV vaccine against infection and intraepithelial neoplasia in women. N Engl J Med. 2015;372(8):711–23.
- 32. U.S. Food and Drug Administration. FDA approves expanded use of Gardasil 9 to include individuals 27 through 45 years old. Content current as of 5 Oct 2018. https://www.fda.gov/news-events/pressannouncements/fda-approves-expanded-use-gardasil-9-includeindividuals-27-through-45-years-old. Accessed 25 Aug 2020.
- 33. Meites E, Szilagyi PG, Chesson HW, Unger ER, Romero JR, Markowitz LE. Human papillomavirus vaccination for adults: updated recommendations of the advisory committee on immunization practices. MMWR Morb Mortal Wkly Rep. 2019;68(32):698–702.
- Giuliano AR, Palefsky JM, Goldstone S, Moreira ED Jr, Penny ME, Aranda C, et al. Efficacy of quadrivalent HPV vaccine against HPV Infection and disease in males. N Engl J Med. 2011;364(5):401–11.
- Drolet M, Benard E, Perez N, Brisson M, Group HPVVIS. Population-level impact and herd effects following the introduction of human papillomavirus vaccination programmes: updated systematic review and meta-analysis. Lancet. 2019;394(10197):497–509.
- 36. Frazer IH, Cox JT, Mayeaux EJ Jr, Franco EL, Moscicki AB, Palefsky JM, et al. Advances in prevention of cervical cancer and other human papillomavirus-related diseases. Pediatr Infect Dis J. 2006;25(2 Suppl):S65–81. Quiz S2.
- World Health Organization. Global Advisory Committee on Vaccine Safety statement on the continued safety of HPV vaccination. 2014. http://www.who.int/vaccine_safety/committee/topics/ hpv/GACVS_Statement_HPV_12_Mar_2014.pdf. Accessed 25 Aug 2020.
- 38. Angelo MG, David MP, Zima J, Baril L, Dubin G, Arellano F, et al. Pooled analysis of large and long-term safety data from the human papillomavirus-16/18-AS04-adjuvanted vaccine clinical trial programme. Pharmacoepidemiol Drug Saf. 2014;23(5):466–79.
- 39. Stokley S, Jeyarajah J, Yankey D, Cano M, Gee J, Roark J, et al. Human papillomavirus vaccination coverage among adolescents, 2007–2013, and postlicensure vaccine safety monitoring, 2006-2014--United States. MMWR Morb Mortal Wkly Rep. 2014;63(29):620–4.
- Moreira ED Jr, Block SL, Ferris D, Giuliano AR, Iversen OE, Joura EA, et al. Safety profile of the 9-valent HPV vaccine:

a combined analysis of 7 phase III clinical trials. Pediatrics. 2016;138(2):e20154387.

- 41. Walker TY, Elam-Evans LD, Yankey D, Markowitz LE, Williams CL, Mbaeyi SA, et al. National, regional, state, and selected local area vaccination coverage among adolescents aged 13–17 years United States, 2017. MMWR Morb Mortal Wkly Rep. 2018;67(33):909–17.
- 42. Bruni L, Diaz M, Barrionuevo-Rosas L, Herrero R, Bray F, Bosch FX, et al. Global estimates of human papillomavirus vaccination coverage by region and income level: a pooled analysis. Lancet Glob Health. 2016;4(7):e453–63.
- Ward K, Quinn H, Bachelor M, Bryant V, Campbell-Lloyd S, Newbound A, et al. Adolescent school-based vaccination in Australia. Commun Dis Intell Q Rep. 2013;37(2):E156–67.
- Shapiro GK, Guichon J, Kelaher M. Canadian school-based HPV vaccine programs and policy considerations. Vaccine. 2017;35(42):5700–7.
- Kane MA, Serrano B, de Sanjose S, Wittet S. Implementation of human papillomavirus immunization in the developing world. Vaccine. 2012;30(Suppl 5):F192–200.
- Pan American Health Organization. PAHO Revolving Fund 2017. https://www.paho.org/en/resources/paho-revolving-fund. Accessed 2 May 2020.
- Vandelaer J, Olaniran M. Using a school-based approach to deliver immunization-global update. Vaccine. 2015;33(5):719–25.
- 48. Fregnani JH, Carvalho AL, Eluf-Neto J, Ribeiro Kde C, Kuil Lde M, da Silva TA, et al. A school-based human papillomavirus vaccination program in Barretos, Brazil: final results of a demonstrative study. PLoS One. 2013;8(4):e62647.
- Peirson L, Fitzpatrick-Lewis D, Ciliska D, Warren R. Screening for cervical cancer: a systematic review and meta-analysis. Syst Rev. 2013;2:35.
- Sankaranarayanan R, Nene BM, Shastri SS, Jayant K, Muwonge R, Budukh AM, et al. HPV screening for cervical cancer in rural India. N Engl J Med. 2009;360(14):1385–94.
- Papanicolaou GN, Traut HF. The diagnostic value of vaginal smears in carcinoma of the uterus. 1941. Arch Pathol Lab Med. 1997;121(3):211–24.
- 52. Wingo PA, Cardinez CJ, Landis SH, Greenlee RT, Ries LA, Anderson RN, et al. Long-term trends in cancer mortality in the United States, 1930–1998. Cancer. 2003;97(12 Suppl):3133–275.
- 53. Huh WK, Ault KA, Chelmow D, Davey DD, Goulart RA, Garcia FA, et al. Use of primary high-risk human papillomavirus testing for cervical cancer screening: interim clinical guidance. Gynecol Oncol. 2015;136(2):178–82.
- 54. Monsonego J, Cox JT, Behrens C, Sandri M, Franco EL, Yap PS, et al. Prevalence of high-risk human papilloma virus genotypes and associated risk of cervical precancerous lesions in a large U.S. screening population: data from the ATHENA trial. Gynecol Oncol. 2015;137(1):47–54.
- 55. Cuzick J, Clavel C, Petry KU, Meijer CJ, Hoyer H, Ratnam S, et al. Overview of the European and North American studies on HPV testing in primary cervical cancer screening. Int J Cancer. 2006;119(5):1095–101.
- 56. WHO. Comprehensive cervical cancer control: a guide to essential practice. 2nd ed. Geneva: WHO; 2014. https://www.who.int/ reproductivehealth/publications/cancers/cervical-cancer-guide/en/. Accessed 25 Aug 2020.
- Jeronimo J, Castle PE, Temin S, Denny L, Gupta V, Kim JJ, et al. Secondary prevention of cervical cancer: ASCO resource-stratified clinical practice guideline. J Glob Oncol. 2017;3(5):635–57.
- 58. Ogilvie GS, van Niekerk D, Krajden M, Smith LW, Cook D, Gondara L, et al. Effect of screening with primary cervical HPV testing vs cytology testing on high-grade cervical intraepithelial

neoplasia at 48 months: the HPV FOCAL Randomized Clinical Trial. JAMA. 2018;320(1):43–52.

- 59. Saslow D, Solomon D, Lawson HW, Killackey M, Kulasingam SL, Cain JM, et al. American cancer society, American society for colposcopy and cervical pathology, and American society for clinical pathology screening guidelines for the prevention and early detection of cervical cancer. J Low Genit Tract Dis. 2012;16(3):175–204.
- United States Preventive Services Task Force, Curry SJ, Krist AH, Owens DK, Barry MJ, Caughey AB, et al. Screening for cervical cancer: US preventive services task force recommendation statement. JAMA. 2018;320(7):674–86.
- Hilal Z, Tempfer CB, Burgard L, Rehman S, Rezniczek GA. How long is too long? Application of acetic acid during colposcopy: a prospective study. Am J Obstet Gynecol. 2020; https://doi. org/10.1016/j.ajog.2020.01.038.
- 62. Massad LS, Einstein MH, Huh WK, Katki HA, Kinney WK, Schiffman M, et al. 2012 Updated consensus guidelines for the management of abnormal cervical cancer screening tests and cancer precursors. J Low Genit Tract Dis. 2013;17(5 Suppl 1):S1–S27.
- Martin-Hirsch PP, Paraskevaidis E, Bryant A, Dickinson HO. Surgery for cervical intraepithelial neoplasia. Cochrane Database Syst Rev. 2013;(12):CD001318.
- Small W Jr, Bacon MA, Bajaj A, Chuang LT, Fisher BJ, Harkenrider MM, et al. Cervical cancer: a global health crisis. Cancer. 2017;123(13):2404–12.
- 65. Moretti-Marques R, Salcedo MP, Callegaro Filho D, Lopes A, Vieira M, Fontes Cintra G, et al. Telementoring in gynecologic oncology training: changing lives in Mozambique. Int J Gynecol Cancer. 2020;30(1):150–1.
- 66. Salcedo MP, Lorenzoni C, Schmeler KM. Working together to eliminate cervical cancer: a partnership across three countries "As mudanças no mundo são criadas por nós". Int J Gynecol Cancer. 2019; https://doi.org/10.1136/ijgc-2019-000372.
- 67. Shastri SS, Mittra I, Mishra GA, Gupta S, Dikshit R, Singh S, et al. Effect of VIA screening by primary health workers: randomized controlled study in Mumbai, India. J Natl Cancer Inst. 2014;106(3):dju009.
- Adsul P, Manjunath N, Srinivas V, Arun A, Madhivanan P. Implementing community-based cervical cancer screening programs using visual inspection with acetic acid in India: a systematic review. Cancer Epidemiol. 2017;49:161–74.
- McCaig LF, Nawar EW. National hospital ambulatory medical care survey: 2004 emergency department summary. Adv Data. 2006;(372):1–29.

- Gindi RM, Black LI, Cohen RA. Reasons for emergency room use among U.S. adults aged 18–64: national health interview survey, 2013 and 2014. Natl Health Stat Report. 2016;90:1–16.
- Ragin DF, Hwang U, Cydulka RK, Holson D, Haley LL Jr, Richards CF, et al. Reasons for using the emergency department: results of the EMPATH Study. Acad Emerg Med. 2005;12(12):1158–66.
- 72. Weber EJ, Showstack JA, Hunt KA, Colby DC, Callaham ML. Does lack of a usual source of care or health insurance increase the likelihood of an emergency department visit? Results of a national population-based study. Ann Emerg Med. 2005;45(1):4–12.
- Horwitz LI, Green J, Bradley EH. US emergency department performance on wait time and length of visit. Ann Emerg Med. 2010;55(2):133–41.
- 74. Llovera I, Ward MF, Ryan JG, LaTouche T, Sama A. A survey of the emergency department population and their interest in preventive health education. Acad Emerg Med. 2003;10(2):155–60.
- Cummings GE, Francescutti LH, Predy G, Cummings G. Health promotion and disease prevention in the emergency department: a feasibility study. CJEM. 2006;8(2):100–5.
- Rosenthal SL, Weiss TW, Zimet GD, Ma L, Good MB, Vichnin MD. Predictors of HPV vaccine uptake among women aged 19-26: importance of a physician's recommendation. Vaccine. 2011;29(5):890–5.
- Hill M, Okugo G. Emergency medicine physician attitudes toward HPV vaccine uptake in an emergency department setting. Hum Vaccin Immunother. 2014;10(9):2551–6.
- Arbyn M, Castle PE. Offering Self-Sampling Kits for HPV Testing to reach women who do not attend in the regular cervical cancer screening program. Cancer Epidemiol Biomark Prev. 2015;24(5):769–72.
- 79. Enerly E, Bonde J, Schee K, Pedersen H, Lonnberg S, Nygard M. Self-sampling for human papillomavirus testing among nonattenders increases attendance to the Norwegian cervical cancer screening programme. PLoS One. 2016;11(4):e0151978.
- Montealegre JR, Landgren RM, Anderson ML, Hoxhaj S, Williams S, Robinson DJ, et al. Acceptability of self-sample human papillomavirus testing among medically underserved women visiting the emergency department. Gynecol Oncol. 2015;138(2):317–22.
- Adler D, Abar B, Wood N, Bonham A. An intervention to increase uptake of cervical cancer screening among emergency department patients: results of a randomized pilot study. J Emerg Med. 2019;57(6):836–43.

e-mail: msandoval@mdanderson.org

Department of Emergency Medicine, Division of Internal Medicine, The University of Texas MD Anderson Cancer Center,

M. Sandoval (\boxtimes)

Houston, TX, USA

Marcelo Sandoval

Introduction

Lung cancer (LC) is the most common and fatal cancer worldwide; in the USA incidence is decreasing, but it remains the deadliest cancer. Patients with LC come to the emergency department (ED) for disease or treatment complications. New LC will be discovered in some, and those with lung cancer diagnosed in the ED are more likely to have advanced disease refractory to treatment. Lesions that are likely benign with a small chance of representing early cancer are sometimes found in the course of ED evaluation. The current standard of care is to refer these patients for followup as recommended by various guidelines.

The historically grim prognosis for LC highlights the need for early detection. Lung cancer screening (LCS) of high-risk asymptomatic individuals with low-dose computed tomography (LDCT) has been shown to reduce mortality by finding LC at earlier stages but at the cost of false positives that can lead to unnecessary distress, radiation exposure, and invasive procedures. As the evidence base for screening grew, steps to minimize false positives and unintentional harm have been taken. Currently only a fraction of eligible patients are screened even though the Centers for Medicare and Medicaid Services (CMS) approved LDCT screening for LC as a preventative health service at 100% coverage in 2015. Currently, many patients at high risk for LC are likely coming to the ED for other reasons. We will review current evidence regarding LCS benefits and harms, barriers to screening, and how the new field of oncologic emergency medicine may help meet currently unmet LCS needs.

> A 65-year-old "frequent flyer" who you see several times a month for non-emergent problems approaches you. "Hey Doc. I just came by to tell you I just turned 65, and I actually

Case Studies

A 65-year-old woman presents with cough, chest pain, and shortness of breath. A CT pulmonary angiogram shows no evidence of pulmonary embolism; however, it reveals a left upper lobe infiltrate, left pleural effusion, enlarged hilar lymph nodes, and several lobulated lesions in the left lung of varying sizes; largest one is about 35 mm, in the area where her pneumonia is present. On further questioning she admits to weight loss and hemoptysis over the past 2 months. After full evaluation you judge she is stable to go home with oral antibiotics. However, you inform her that some of the findings on history and CT are highly suggestive of LC. "How can that be? I've never smoked in my life! What are my chances, doctor?" as she begins to cry. What can you tell her about how she might have developed this, and what is her prognosis?

A 45-year-old male with upper abdominal pain and yomiting and tenderness in the epigastrium (but no Murphy's or peritoneal signs) is found to have an elevated lipase, total and direct bilirubin, and alkaline phosphatase. He has a history of fatty food intolerance and denies drinking alcohol. CT abdomen confirms a dilated common bile duct and mild pancreatic inflammation due to a retained common bile duct gallstone. He feels much better after symptom relief and IV fluids. Given the absence of fever or WBC elevation, you decide to discharge him with GI follow-up the next day. Fifteen minutes later the radiologist calls you. "Hi there. I looked more closely, and he also has a right lower lobe lung nodule that's 10 mm in diameter, with subsolid/ground glass and part-solid features, minimal attenuation of surrounding structures, and no calcifications. I thought you should know." He hangs up and you ask yourself: "Hmm, do I even need to tell him about this? It doesn't sound so big and he has no chest symptoms. If it's nothing, I don't want to worry him. What should I do?"

12



161

[©] Springer Nature Switzerland AG 2021

K. H. Todd et al. (eds.), Oncologic Emergency Medicine, https://doi.org/10.1007/978-3-030-67123-5_12

got my Medicare card vesterday, so now I don't have to come see you so often anymore! Maybe you can tell me how to get my hernia fixed I've been putting off, and seeing somebody for my knee pains and back pains instead of having to come see you all the time?" You smile because you like this patient, and after seeing him so often over the years, you've actually developed a rapport with him. You know he needs primary care. If only he would stop smoking! You tried to enroll him in smoking cessation a few times, but he wasn't interested. He told you "I feel good, I want to enjoy my life, and so what's the point of quitting? If I get the big C, there's nothing they can do anyway. My friend started coughing last year, he went to the doctor, and they found it in his lung. Two months later he was dead." Is he right? You know LC is very bad, and almost no one gets diagnosed early enough for surgical resection. So is there any point in looking for trouble since he feels well and looks pretty good for someone his age?

Background

LC is the most common and fatal cancer worldwide with 2.1 million new cases and 1.8 million deaths recorded in 2018 [1, 2]. The USA will see 229,000 new cases in 2020 [3]. Prostate and breast cancer are more common in men and women, respectively, but LC will claim 136,000 Americans, more than colon, prostate, and breast combined [3]. The National Cancer Institute (NCI) estimated LC expenditure has risen to \$14.2 billion in 2018. In terms of lost productivity, \$36.1 billion dollars are lost per year due to LC, more than for any other cancer [4].

Cigarettes and smoke exposure account for 90% of LC [5–7]. Pre-existing lung disease such as COPD and pulmonary fibrosis increases cancer risk and mortality. Smokers with COPD die from LC more than smokers without COPD [8, 9]. Cannabis-related risk is uncertain; pooled analysis in 2015 showed no specific links, but the investigators stated that they did not include enough heavy users [10]. More potent strains, medical and recreational use make further study warranted. Data on hookah smoking [5] and electronic cigarette-related LC risk [5, 7] are very sparse.

Never smokers (<100 cigarettes/lifetime) account for 15–25% of LC. Very common among Asian women [11, 12], it is rising in the USA [13] and affects women more than men for reasons that include lifestyle, genetic, and hormonal differences [14–16].

Environmental risk factors include uranium-derived radon in soil, homes [17], air pollution [18–21], and indoor coal burning [22]. Occupational risks include asbestos, arsenic, beryllium, cadmium, chromium, diesel fumes, nickel, and silica [5, 7, 11, 23]. Other known risk factors are age, HIV and family history, *Mycobacterium tuberculosis*, human papillomavirus, low socioeconomic status (SES), and Black, non-Hispanic White, or Native American status [5, 7, 11, 23].

Epidemiology and High-Risk Populations

Lung cancer occurs decades after smoking, so populationwide changes in incidence (new cases per 100,000 per year) and mortality (deaths per 100,000 per year) also lag by decades behind changes in smoking rates. Africa has the lowest rates of smoking [24], but use is increasing as disposable income grows [25]. Smoking peaked earliest among males (1970s) and females (1990s) in the USA, UK, Canada, and Australia; because men and women have been smoking less for decades, LC incidence and mortality have been declining for years in these countries [6, 7, 23, 26]. Higher-income northwestern European, South American, and Asian populations are experiencing more recent declines in incidence and mortality among males, due to decreases in male smoking over the last two decades [23]. At the same time, women's mortality is rising due to prior increases in smoking [6]. In low- and middle-income populations of southeast Europe, Asia, and South America, LC is rising sharply in men, less so in women, accounting for the recent global increases in cancer [6, 7, 26]. Air pollution also contributes to these rising rates, especially in China and Russia [27]. Among women, incidence and mortality rates have been declining in North America, Western Europe, Australia, and East Asia but remain the highest worldwide in these places [7, 28]. On a positive note, the World Health Organization's (WHO) Cancer Mortality Database showed tobacco cessation in North America, Europe, and South America has led to decreases in LC incidence and mortality in those under age 50 [29].

In the USA, despite incidence and mortality declines in both men and women, certain populations remain at high risk. 2019 American Cancer Society statistics show that Black Americans have long had the highest incidence and mortality of all ethnic/racial groups, despite declines in smoking among Black teens since the 1970s [30]. Incidence and mortality in Black individuals are followed closely by non-Hispanic Whites and Native Americans [31]; each group has about approximately 60-65 per 100,000 incidences and 35-45 per 100,000 mortalities. Black American male LC incidence (85.4) and mortality (63.9) are much higher than Black women (49.2 and 33.3, respectively). Men account for most of the burden among non-Hispanic whites and Native Americans as well [31]. Lower socioeconomic status [32] and education level also predict increased incidence and mortality [7]. A LC mortality belt across rural counties in Oklahoma, Arkansas, southern Missouri, southern Illinois, Kentucky, Tennessee, southern Ohio, West Virginia, and western Virginia is highly correlated with poverty, unemployment, lack of health insurance, obesity, and low air quality [21], as well as increased rates of smoking and decreased surgical treatment of stage I disease [33]. Asian and Hispanic Americans of both genders have the least incidence and mortality [6, 7, 31], and a 2019 meta-analysis showed that being female (any group), Hispanic, or Asian American predicted increased survival from LC [34].

Others at risk for LC are lesbian, gay, bisexual, and transgender (LGBT) people; the disabled; mentally ill individuals; and low pay grade military personnel, due to disproportionate rates of smoking [35].

Pathology, Prognosis, and Treatment

80% of LCs are non-small cell (NSCLC), and 15% are small cell LCs (SCLCs); other carcinoid/neuroendocrine tumors account for the remainder [36–38].

The most common NSCLC is small-airway adenocarcinoma followed by central large-airway-based squamous cell carcinoma (SCC), which is almost always associated with smoking [39]. Traditionally, NSCLC referred to both as treatment was similar. However, since 2015 WHO guidelines emphasize biomarkers for individualized treatment with targeted and immunotherapy regimens [38, 40] making differentiation more important. In the early stage, both are amenable to resection.

Since the 1990s, adenocarcinoma is increasing, due to filtered cigarettes (allowing small carcinogens to penetrate deeper into the airways), while decreases in smoking have reduced SCC [23]. "Never smoker" LC is usually adenocarcinoma and is more common in women [15].

SCLC, also associated with smoking, is highly aggressive and classified as limited and extensive, with 20–25% 5-year survival and 10% 2-year survival, respectively [41]. There is hope combinations of chemotherapy and immunotherapy can improve this [42].

Early diagnosis and surgical cure has long been key to best prognosis [43–45], with stage I 5-year survival rates of 45–79% [46–48]. Efforts to improve outcomes from curative surgery have focused on techniques such as video-assisted thoracic surgery lobectomy (VATS), sublobar resection (SLR) [48], prognostic markers [49], and pre-surgical adjunctive treatments with immunotherapy [50]. For advanced LC, prognosis has changed little even with newer treatments [51, 52]. According to the NCI Surveillance, Epidemiology, and End Results Program (SEER), 5-year survival with localized disease is about 59% but only 5.8% in distant metastatic disease. Unfortunately, only 17% present with localized disease [3].

Lung Cancer in the Emergency Department

Data from the Nationwide Emergency Department Sample (NEDS) showed that of nearly 5 million US cancer-related ED visits in 2012, 10.3% were by known LC patients [53]. These patients most frequently went to EDs for pneumonia,

COPD, and respiratory failure with 67% of visits resulting in admission. In a 2019 prospective observational cohort study of 1075 ED cancer patients, of whom 139 (12.9%) were diagnosed with LC, the most common symptoms were abdominal pain, fever, breathing abnormalities, nausea/vomiting, and throat/chest pain [54]. Major symptoms leading to ICU admission and short-term mortality were shortness of breath and altered mental status [55].

Diagnosis of cancer in the ED is summarized in a 2017 review by Zhou. Such patients are less likely to be curable, less likely to survive at 1 year, 3 months, and 1 month compared to those diagnosed electively, even when adjusting for staging [56]. It is relatively uncommon that emergency physicians expect to diagnose LC in a patient with cardiopulmonary symptoms. Serious illnesses such as pulmonary emboli, pericardial tamponade, acute coronary syndromes, pulmonary edema, anaphylaxis, pneumothorax, pneumonia, COPD exacerbations, and asthma are more commonly expected [57], and, of course, COPD and cardiovascular disease frequently co-exist with LC [58]. Common symptoms leading to LC diagnoses in the ED include cough, shortness of breath, chest pain, and/or hemoptysis [58–64]. Hemoptysis is the most predictive symptom [62-64], even more so if associated with cough, dyspnea, and weight loss [62].

A 2012 Michigan study compared patients diagnosed with LC in the ED to those diagnosed in other settings and found that ED patients had more co-morbidities, were older, more often female, more likely to be Black and more likely to present in advanced stages [65]. The authors noted the need for increased ED smoking cessation and cancer screening awareness.

Urban Black and Hispanic patients face disparities in health care, will go to the ED, and have worse outcomes for many conditions, including cancer [66]. In rural America, White individuals have twice the mortality from LC than their urban counterparts, even with similar staging distributions [33].

ED Imaging and Lung Cancer

Chest X-ray (CXR) has long been used for ED chest complaints [67, 68]. Since the late 1990s, however, computed tomography (CT) has replaced CXR for diagnosis of pulmonary emboli [69, 70], aortic dissection [71], coronary disease [72], and some pneumonias [73, 74] as it is more sensitive and more often changes management [73]. ED CT scanning also discovers more masses [75–77] and nodules [78, 79] than does plain CXR.

Lung masses (>30 mm) [80] are likely malignant; thus prompt referral to a pulmonologist is recommended. Other CT findings highly suggestive of cancer include pleural effusion, pleural nodules, enlarged hilar or tracheal nodes, endobronchial lesions, post-obstructive pneumonia, spiculated or lobular borders, presence of a solid component within a ground glass lesion, and growth on serial imaging [81].

Nodules are rounded or irregular opacities (<30 mm) [80]. Attenuation can be solid (opaque) or subsolid, which can further be classified as ground glass (hazy) or part-solid (mixed features) [80]. Solid nodules are less likely to be malignant, especially if calcified or <6 mm in size, unless they are new or growing. Subsolid nodules are also unlikely to be malignant if smaller than 6 mm. Factors that increase the likelihood of malignancy include size >15 mm, spiculation, lobulation, upper lobe location, presence of COPD or pulmonary fibrosis, and a history of cancer or growth. Multiple (two to four) nodules are more likely malignant, but five or greater are less likely (unless there is history of cancer). The Fleischner Society makes recommendations for follow-up of incidental nodules found on CT. Unfortunately, discovery of these incidental nodules often happens after the patient has left the ED [82, 83], at which point the ED has responsibility for notifying the patient. Follow-up recommendations may vary from none to 3-12 months later, depending on the size, consistency, rate of growth, and patient factors. These guidelines do not apply to nodules found on CT screening for cancer (see Lung-RADS[®] below) or if the patient is < age 35 or immunocompromised or has a history of cancer [84]. For incidental nodules found in the ED, use of Fleischner Society Guidelines decreases the rate of follow-up scans and invasive procedures [85].

Lung Cancer Prevention

Smoking bans and cessation decrease LC incidence rates in a predictable fashion [86]. Cessation before age 40 reduces smoking-related mortality by 90%. Smoking cessation efforts in ED are fully discussed in another chapter of this book.

Fine particulate matter air pollution (PM2.5) is increasingly recognized as a risk factor for LC, disproportionately in the South and Midwest, among Black individuals and those living in socioeconomically deprived areas. Much of the risk associated with fine particulates occurs below current Environmental Protection Agency (EPA) standards for PM2.5, suggesting the need to reassess current clean air standards in the USA [87].

Although home radon testing is recommended by the CDC and EPA, it is infrequently performed as no state currently requires home testing. Some states mandate radon testing in public buildings or schools. Efforts to increase low testing rates include novel approaches such as smartphone apps [88]. Targeted public health efforts to increase radon testing in areas (Kentucky) where both smoking rates and radon levels are high [89] may be particularly effective given the known synergy of radon and smoking in causing LC.

Lung Cancer Screening with Chest X-Ray and the Problem of Overdiagnosis

LC screening (LCS) attempts early detection before symptoms appear, thus improving prognosis. The NCI-sponsored 1986 Mayo Lung Project randomized 2 groups of approximately 4600 male smokers to screening CXRs and sputum cytology every 4 months for 6 years against standard care (recommended annual CXR and sputum cytology). Twice as many early cancers were found, more surgery was done, and 5-year LC-specific survival was superior in the screening group; however, there was no benefit in overall mortality over 6 years [90]. Other contemporaneous NCI trials at Memorial Sloan Kettering [44] and Johns Hopkins [91] also confirmed lack of mortality benefit.

It is important to understand the difference between survival and mortality to interpret these seemingly contradictory results. Survival is measured as the percentage of those diagnosed with cancer alive after time, traditionally 5 years. It can be helpful when telling a patient how long they may have to live but is also used to identify improvements in treatment. Mortality is measured as those who die from a given disease per 100,000 of the general population per year. For a treatment to be deemed effective, both survival and mortality should improve [92]. Either can be increased by preventing death (screening or treatment) or preventing disease (smoking cessation), but survival can also be increased by diagnosing more people earlier, some of whom will not get sick. This phenomenon is intrinsic to cancer screening [93-95] and is termed overdiagnosis - i.e., finding early indolent (slowly progressive) cancers that will not progress to symptoms or death. The Mayo Lung Project found 206 cancers in the screening group and only 160 in the control group over 6 years; more early cancers were found by screening, but roughly equal numbers of late cancers were found in both groups [90]. If a randomized controlled trial is properly designed, and screening finds more early cancers, it should also find fewer late cancers (stage shift), keeping total incidence in the two groups approximately equal [94–96]. In addition, the smaller non-screening control group should eventually "catch up" to the screening group in numbers of cases, since two properly matched cohorts should have the same incidence of cancer over time. Too few subjects or too short a study period will not allow a stage shift or catchup to be detected.

Technology that increases early detection will increase incidence (by finding more fledgling cases) and will also appear to improve survival because some of these will be slow-growing lesions that are not destined to harm over the follow-up period. Mortality, a static measure of deaths per year, is not affected by increases in detection of new cases, so it is not subject to overdiagnosis [92]. Over time, a relative reduction in mortality between the two groups is the best way to see if there is benefit to the treatment or screening.
The initial Mayo mortality rate over 6 years was equal – 122 deaths in the screening group and 115 deaths in the control group; 3.2 vs. 3.0 (deaths per 1000 person-years). When follow-up extended to 16 more years, there were still more cancers (585 vs. 500) in the screening group, but no difference in mortality (4.4 vs. 3.9 death/person-years). This strongly suggests overdiagnosis of indolent lesions in the screening arm that never progressed because these cancers did not result in more deaths [97]. The price of overdiagnosis is inappropriate testing and treatment of these indolent cancers.

Overdiagnosis has been long recognized in prostate and breast cancer [98]. The existence of indolent LCs was known for some time from autopsy studies [99] but was thought to be uncommon because most people who developed LC died. Formerly, these indolent cancers were known as bronchoalveolar carcinoma (BAL) but since 2015 have been reclassified by WHO and the American Joint Committee on Cancer (AJCC) [40, 100]. These lesions appear as malignancies on radiography and biopsy, and as there are no metastases, they are usually resected. However on resection they are found to have a lepidic growth pattern and are termed adenocarcinoma in situ (AIS) or minimally invasive adenocarcinoma (MIA). Both are very slow growing with almost 100% 5-year survival after resection compared to only 67% with other types on adenocarcinoma. They are so indolent that they may never manifest before a patient dies for other reasons, but identifying these prior to resection is not yet routine [100]. NCI is sponsoring research in genomic and molecular techniques to detect them, allowing for less invasive treatment or surveillance leading to similar good outcomes [101]. Since the advent of CT scanning, there is evidence that this occurs more frequently than previously thought, in up to 18% of cancers detected using low-dose computed tomography (LDCT) [102, 103].

The abovementioned CXR studies may have been underpowered to determine whether screening might have a small mortality benefit. The 2011 Prostate, Lung, Colorectal and Ovarian (PLCO) Cancer Screening Trial was designed to detect smaller differences in mortality, randomizing 155,000 healthy smoking and non-smoking volunteers to annual CXR screening vs. usual care for 3 years. There was no mortality benefit [104], thus putting the question of CXR screening to rest.

Computed Tomography (CT) for Lung Cancer Screening

In the 1970s and 1980s, the potential of CT scanning for LCS led to a 1998 Japanese study by Sone et al. [105] and the 1999 Early Lung Cancer Action Project (ELCAP) study by Henschke et al. [106]. Both studies used LDCT protocols

which had been developed and studied by Naidich [107] and Kaneko [108] with lower radiation doses than standard highresolution scanning. Both found LDCT was vastly superior to CXR in finding earlier and smaller cancers; however, potential overdiagnosis, unnecessary testing, and invasive procedures made the value of CT screening less clear [94, 95, 109]. In 2006, Henschke et al. published a larger international multi-center study (I-ELCAP) evaluating annual screening CTs of smokers and non-smokers with other LC risks and incorporating 10- rather than standard 5-year follow-up. 412 of 484 detected cancers were in clinical stage I, with 88% 10-year survival with treatment; 92% if surgery was done within 1 month. The study concluded that CT screening could detect curable LCs as evidenced by longterm follow-up results [110]. Once more, the problem of overdiagnosis was cited [111] and the possibility that early AIS could regress due to innate defense mechanisms [112]. Also, as a one-arm observational case series without a control group, there could be no conclusion that early detection decreases mortality, despite the impressive 10-year survival rate. A randomized controlled study comparing CT screening with standard care was sorely needed [112].

National Lung Screening Trial

With this background, the NCI-funded National Lung Screening Trial (NLST) was initiated in 2002. Over 2 years, 53,454 current or former smokers aged 55-74 with 30 packyears of smoking were recruited; if not currently smoking, participants had quit less than 15 years before. The subjects were randomized to undergo either three annual screening CXRs or low-dose CTs, and LC incidence and mortality were recorded for a mean of 6.5 years. The results were published in 2011. For the first time, screening showed a relative mortality reduction of 20% for LC and 6.7% in all-cause mortality. Unlike the CXR studies, "stage shift" was seen, more stage I (50% vs. 31%) but less stage IV (22% vs. 36%) cancers. Three hundred fifty-six deaths from LC occurred (247 per 100,000 person-years) compared with 443 deaths (309 per 100,000 person-years) in the chest radiography group. The number needed to screen (NNS) to save one life was 320. This was a landmark study demonstrating mortality reduction with LCS for the first time [113].

However, there were several concerns. Silvestri [114] and Bach et al. [115] noted that ~25% of CTs each year found "positive" nodules of which 96% were ultimately benign. Most follow-up for the "positive" nodules was radiographic, but approximately 6% had invasive testing (73% negative malignancy rate), and 10% had surgery (21% negative for cancer). Six patients without cancer died within 60 days due to workups. Furthermore, the study had well-trained radiologists and surgeons (surgical death rate was only 1%, much better than average), motivated volunteer subjects, mostly ex-smokers, and only 8% were over age 70. Hospitals with less compliant, older patients with more co-morbidities and less experienced physicians would likely have worse benefit and harm rates. Screening benefits could be offset by increased smoking in those who became anxious from false positive results or falsely reassured by negative ones. Increased radiation and reduced quality of life due to anxiety were also cited [114, 115].

Overdiagnosis again became an issue. Finigan and Kern noted that of 119 additional cancers detected by CT, 75 were indolent bronchoalveolar carcinoma (BAL), now known as AIS and MIA [96]. Patz et al. analyzed NLST data and estimated an 18.5% overdiagnosis rate [102]. In 2013, the American Academy of Family Medicine (AAFP) concluded that insufficient evidence existed to recommend for or against LCS with low-dose CT [116].

The US Preventive Services Task Force (USPSTF) reported their assessment of LCS with low-dose CT in 2013. Humphrey et al. noted in a review written for the USPSTF that the NNS of 320 was better than those for mammography and sigmoidoscopy [117]. Because of concerns regarding overdiagnosis and harm, the Cancer Intervention and Surveillance Modeling Network (CISNET) developed screening simulation models incorporating age, frequency, and smoking history combinations predicting the most benefit for the least CT scans. The models predicted that by increasing age to 55-80 and keeping smoking requirements the same would result in superior outcomes than NLST criteria [118]. The USPSTF made a level B recommendation to Congress that all smokers aged 55-80 years and that have a 30 pack-year smoking history and currently smoke or quit within the past 15 years should get annual LDCT screening [119]; this led CMS (Centers for Medicare and Medicaid Services) to approve LCS as an annual preventative service benefit for all eligible Americans up to age 77 without copayment in 2015 [120].

There are strict requirements for federal LCS coverage and reimbursement [120, 121], including a documented shared decision-making conversation in the medical record that specifies the patient's eligibility by age and smoking history, the risks and harms of screening, and concurrent offers of smoking abstinence and cessation. LC CT screening programs must be accredited and use a central registry: the ACR Lung Cancer Screening Registry (LCSR), provided by the American College of Radiology [122]. In 2014 the ACR released Lung-RADS®, the Lung CT Screening Reporting & Data System [123] to better standardize reporting of findings among radiologists reading screening CT scans. Lung-RADS® increased nodule size threshold to 6 mm and incorporated other evidence-based criteria for nodule management. Applying the criteria to the NLST data, there was a 50–75% decrease in false positives which could lead to less invasive procedures [124]. Lung-RADS[®] is now the standardized method for radiologists to report on LCS nodules and make management recommendations. Version 1.1 was released in 2019 (Fig. 12.1).

Barriers to LDCT Screening

Jemal and Fedewa used the CDC National Health Interview Survey (NHIS) and found less than 4% of eligible Americans underwent LCS in 2015, and about 50% were uninsured or on Medicaid [125]. In comparison, screening rates among eligible candidates is 83% for Pap tests, 71.5% for mammography, and 62.4% for colorectal cancer screening tests [126].

Since 2012 the National Comprehensive Cancer Network (NCCN), comprised of 30 leading cancer centers dedicated to improving cancer care, has added a second group (group 2) to the population that should be eligible for LCS. NCCN consensus is that risk comparable to NLST criteria exists for patients age >50, 20+ pack-years of smoking, and one other risk factor (history of cancer, COPD/pulmonary fibrosis, family history of LC, radon exposure, or occupational exposure to carcinogens). Though initially a lower-level 2B recommendation in 2012 [127], the NCCN upgraded this to 2A in 2015 [128]. Pinsky had written in 2012 that strict adherence to NLST criteria would only find about 27% of lung cancers [129]; if criteria was expanded to those age 50-79, and any smoking history (current or quit <15 years), almost 50% of cancers could be found, though there would be much higher NNS and possible harm. By 2015 he found that the risk of current smokers with 20–29 pack-years was equivalent to that of former smokers with 30 or more pack-years; more women and minorities would be eligible for screening if the smoking requirement was reduced [130]. NCCN has recognized these and other studies by continuing to recommend group 2 LCS in their 2020 guidelines; they do acknowledge "true risks and benefits of screening these group 2 individuals are uncertain" and state a PLCO risk score [131] may help select the most appropriate group 2 individuals for LCS [132]. Two studies at single centers compared LCS in NCCN group 1 and 2 patients and found 1-2% rates of new cancer in both, but neither could not comment on mortality due to short follow-up or small sample size [133, 134]. The American Association for Thoracic Surgery has adopted similar LCS guidelines, advocating NCCN group 2 patients should be eligible for screening. They also added another group: those who have a history of LC and have been disease-free for 4 years [135].

Barriers to LCS in patients are awareness, cost, fear of cancer diagnosis, stigma of smoking, and radiation fears. Providers' barriers include unfamiliarity, lack of prompt access to accurate smoking history, time constraints for

Fig. 12.1 Lung CT Screening Reporting & Data System (Lung-RADS[®]). Lung-RADS® v1.1 Assessment Categories (2019 Release) (Courtesy and with permission of the American College of Radiology, CC BY-ND 4.0)



Lung-RADS[®] Version 1.1

Assessment Categories Release date: 2019

Category Descriptor	Lung- RADS Score	Findings	Management	Risk of Malignancy	Est. Population Prevalence
Incomplete	0	Prior chest CT examination(s) being located for comparison Part or all of lungs cannot be evaluated	Additional lung cancer screening CT imaged and/or comparison to prior chest CT	n/a	1%
Negative		No lung nodules	examinations is needed		
No nodules and definitely benign nodules	1	Nodule(s) with specific calcifications: complete, central, popcorn, concentric rings and fat containing nodules			
	2	Perifissural nodule(s) (See Footnote 11) < 10 mm (524 mm ³)	Coutinue annual screening with LDCT in 12 months	< 1%	90%
Benign Appearance or Behavior		Solid nodule(s): < 6 mm (< 113 mm ³) new< 6 mm (< 34 mm ³)			
Nodules with a very low likelihood of becoming a		Part solid nodule(s): < 6 mm total diameter (< 113 mm ³) on baseline screening			
clinically active cancer due to size or lack of growth		Non solid nodule(s) (GGN): < 30 mm (< 14137 mm ³) OR \geq 30 mm (\geq 14137 mm ³) and unchanged or slowly growing			
		Category 3 or 4 nodules unchanged for ≥ 3 months			
Probably Benign Probably Benign finding(s) - short term follow up suggested; includes noduless with a low likelihood of becoming a clinically active cancer	3	Solid nodule(s): ≥ 6 mm to < 8 mm (≥ 113 to < 268 mm ³) at baseline OR new 4 mm to < 6 mm (34 to < 113 mm ³)	6 month LDCT	1-2%	5%
		Part solid nodule(s): ≥ 6 mm total diameter (≥ 113 mm ³) with solid component < 6 mm (< 113 mm ³) OR new < 6 mm total diameter (< 113 mm ³)			
		Non solid nodule(s): (GGN) ≥ 30 mm (≥ 14137 mm ³) on baseline CT or new			
Suspicious Findings for which additional diagnostic testing is recommended	4A	Solid nodule(s): ≥ 8 to < 15 mm (≥ 268 to < 1767 mm ³) at baseline OR growing < 8 mm (< 268 mm ³) OR new 6 to < 8 mm (113 to < 268 mm ³)	3 month LDCT; PET/CT may be used when there is a ≥ 8 mm (≥ 268 mm ³) solid component	5-15%	2%
		Part solid nodule(s): ≥ 6 mm (≥ 113 mm ³) with solid component ≥ 6 mm to < 8 mm (> 113 to < 268 mm ³) OR with a new or growing < 4 mm (< 34 mm ³) solid component			
		Endobronchial nodule			
Very Suspicious Findings for which additional diagnostic testing and/or tissue sampling is recommended	4B	Solid nodule(s): ≥ 15 mm (≥ 1767 mm ³) OR new or growing, and ≥ 8 mm (≥ 268 mm ³)	Chest CT with or Without contrast, PET/CT and/or tissue sampling depending on the *probability of malignancy and comorbidius. PET/CT may be used when there is a ≥8 mm (≥ 268 mm ³) solid component. For new large nodules that develop on an annual repeat screening CT, a 1 month LDCT may be recommended to address potentially inflectious or inflammatory conditions	> 15% 2	2%
		Part solid nodule(s) with: a solid component ≥ 8 mm (≥ 268 mm ³) OR a new or growing ≥ 4 mm (≥ 34 mm ³) solid component			
	4X	Category 3 or 4 nodules with additional features or imaging findings that increases the suspicion of malignancy			
Other Clinically Significant or Potentially Clinically Significant Findings (non lung cancer)	s	Modifier - may add on to category 0-4 coding	As appropriate to the specific finding	n/a	10%
,		I	L		

IMPORTANT NOTES FOR USE:

1) Negative screen: does not mean that an individual does not have lung cancer

Negative screen: does not mean that an individual does not have lung cancer
 Size: To calculate nodule mean diameter, measure both the long and short axis to one decimal point, and report mean nodule diameter to one decimal point.
 Size: To calculate nodules at first detection, and that grow and reach a higher size category.
 Growth: an increase in size of > 1.5 mm (> 2mm³).
 Exam Alegrapy: each exam should be coded 0-4 based on the nodules(s) with the highest degree of suspicion.
 Exam Alegrapy: each exam should be coded 0-4 based on the nodules(s) with the highest degree of suspicion.
 Exam Alegrapy: each exam should be coded 0-4 based on the nodules(s) with the highest degree of suspicion.
 Exam Alegrapy: each exam should be coded 0-4 based on the nodules(s) with the highest degree of suspicion.
 Exam Alegrapy: each examination and the nodules (at the nodule set is the nodule of the nodule set is the nodule set is the nodule set in the set is the nodule of the nodule is nodule set is suspicing findings that increase the suspicion of lung cancer, such as spliculation, GGN that doubles in size in 1 year, enlarged ymph nodes etc cellograpy of and 4 nodules with smooth margins, an oval, lentiform or triangular shape, and maximum diameter less than 10 mm or 524 mm3 (perifissural nodules) should be classified as category of 2.
 Category 2 and 4 nodules that are unchanged on interval CT should be coded as category 2, and individuals returned to screening in 12 months 13)LDCT: low does cher CT

*Additional resources available at - https://www.acr.org/Clinical-Resources/Reporting-and-Data-Systems/Lung-Rads

*Link to Lung-RADS calculator - https://brocku.ca/lung-cancer-screening-and-risk-prediction/risk-calculators/



required shared decision-making, lack of support by specialists, and skepticism about benefit [136, 137].

Racial barriers to LCS were summarized by Borondy Kitts in 2019. Black individuals have higher rates of cancer and death than Whites but may be less likely to qualify for LCS due to lower smoking rates. She called for CMS to approve NCCN group 2 criteria and expand LCS for Black Americans. To date, however, LCS of NCCN group 2 is not covered by CMS or private insurance. Education, socioeconomic status, lack of primary care, mistrust of providers, and providers' implicit bias are additional barriers to LCS uptake among Blacks [138].

Annangi showed Black Americans have rates of earlyonset LC (age 45–54) 73% higher than Whites [139]. A review of NLST data by Balekian et al. demonstrated ethnic disparities in rates of curative surgery for stage I LC. Black men had a 61% rate of curative surgery for stage I, whereas White men and women and Black women underwent curative surgery in 90% of cases. The reasons for these disparities are unclear [140].

Geographically, the South has 40% of screening-eligible patients, but has the lowest density of ACR-designated screening centers, and a 3.5% screening rate. In contrast, the Northeast has the lowest eligible percentage (15.5%) but the highest density and screening rate (10.1%) [141].

23% of the screening-eligible candidates live in rural areas [141]; however, most screening centers are in metropolitan areas; 28% of eligible patients have to drive 30 minutes or more to reach a screening center [142].

LCS barriers in the community hospital setting include low rates of screening by poorer eligible patients in favor of inappropriate screening of ineligible but affluent people, poor adherence to Lung-RADS[®] follow-up recommendations, poor patient compliance with follow-up, and inaccurate smoking data [143].

Huo et al. compared CDC NHIS data from 2010 and 2015, finding a trend toward increased CT LCS by people ineligible for screening and raising concerns regarding overuse of the modality in some settings [144].

While gays, lesbians and bisexuals have higher ratios of smoking, and thus higher percentages who are screeningeligible, they receive LCS ratios rates similar to their heterosexual counterparts [145]. One report from Japan demonstrated that screening-eligible schizophrenics, despite higher smoking rates, are less likely to receive screening [146].

Radiation risk from medical procedures is the subject of another chapter in this book, but perceptions of radiation risk among patients and providers bear discussion. Ionizing radiation is typically quantified as effective dose measured in millisieverts (mSv). LDCT in NLST delivered a mean effective dose of 1.4 mSv [147]. For perspective, the background radiation effective dose is 3 mSv per year, while a coast to coast flight is 0.035 mSv [148]. One CXR = 0.1 mSv per

exam; a standard chest CT = 8 mSv; positron emission tomographic (PET) scans = 14 mSv; and a CT angiogram for pulmonary embolism = 10 mSv [149]. Bach in 2012 pointed out that NLST could lead to 1 radiation-related cancer death per 2500 people screened due to cumulative follow-up CT and PET-CT doses of 8 mSv over 3 years [115]. McCunney calculated that doses to smokers in annual LCS programs for 30 years would exceed that of nuclear plant workers [150]. The risk of developing cancer from LDCT was studied by the investigators of the Italian COSMOS; in their group of 5200 receiving annual CTs for 10 years plus PET-CTs for positive nodules, 1.5 LCs and 2.4 cancers could have occurred. They calculated 1 radiation-induced cancer for every 108 LCs found, but younger women 50-55 had a higher risk of radiation-induced cancer than other groups. They noted that all scans (including follow-up) were performed using low-dose technique, unlike NLST. They stated that Lung-RADS® would further reduce radiation and that technology would reduce CT dose even more; thus radiationassociated toxicity could decrease in time [151].

Proposed Solutions to LCS Barriers

Solutions to improve rates of LCS include provider incentives, use of dedicated navigators for patients and providers, further identification of barriers, and obtaining more accurate smoking histories [136, 137, 143]. Solutions addressing mistrust of the medical system include campaigns to engage patients, using testimonial success stories, avoiding scare tactics, and recruiting trusted community providers (rather than screening program representatives) to invite candidates to screening [136–138]. Racial and socioeconomic disparity solutions include expanding screening to Black Americans using NCCN group 2 criteria [138], seeking funding for free CT screening programs, providing free transportation, mobile CT screening centers, scheduling screening on evenings and weekends to minimize lost time at work [137], and developing culturally tailored educational and outreach programs in collaboration with community health centers and leaders [138].

Successful implementation of LCS program in nonacademic community centers was demonstrated in two recent studies. Over the first 5 years of a community-based LCS program, 1241 patients underwent screening for 1 round, and 29 cancers were found, 72% in stage I. There was a 21.3% Lung-RADS[®] "positive" CT rate, with 20% of these false positive. An unfavorable finding was low adherence (37%) to repeat scanning in those with a negative scan, leading to speculation that this first negative result may falsely reassure some participants [152]. In another larger study, 3400 patients in a large community hospital network were screened over several years, 500 (14.6%) had "positive" CT, and of these 111 had cancer (95 lung, 16 others). Of the 95 LCs, 67 were stage I or II and they reported a 2.3% surgical mortality rate. The program had an 83% adherence rate to repeat screening. Diagnostic and complication rates were comparable to NLST, contradicting the belief that NLST results could not be generalized to the community setting [153]. Future studies may influence the AAFP to change their stance on LCS.

Ultra-low-dose CT (ULDCT) is a newer modality that provides comparable resolution to LDCT, with even lower doses of radiation (~0.13 mSv) almost equivalent to 1 CXR [154]. It is expected that this form of CT scanning will replace LDCT over time, further reducing radiation risk.

Management of nodules found through LCS has evolved since NLST. Lung-RADS[®] criteria raised the positive nodule threshold to 6 mm on initial scan or 4 mm for new nodules on subsequent scans. Growth is defined as an increase of 1.5 mm in size. Criteria were designed to decrease unnecessary repeat CTs, PET scans, or biopsies.

It is possible to reduce the need for repeat CT scans of nodules found on LCS by using software to calculate volumetric size rather than linear diameter to more accurately assess growth and increasing the interval for subsequent scans after the first two annual rounds in those with negative results to 2 years. These ideas were incorporated into subsequent large randomized control trials [155].

The Nederlands-Leuvens Longkanker Screenings Onderzoek (NELSON) Trial

The Dutch-Belgian NELSON study was initiated in 2003 and was designed to find a 25% reduction in mortality with 80% power over 10 years [156]. They recruited adults aged 50-75 with >15 cigarettes per day for >25 years or >10 cigarettes per day for >30 years, currently smoking or had quit <10 years previously. They were randomized to screening vs. standard care without screening. The protocol called for a baseline CT, subsequent scans at 1, 3, and 5.5 years and then follow-up for 10 years. The most obvious differences between NELSON and NLST were the lower age limit and less intense smoking history (mean 38 pack-years vs. 56 pack-years NLST) requirement. By including subjects with moderate risk, the investigators calculated that they be would be able to recruit the necessary number of people (16,000) to adequately power the study, as well as apply the results to a broader population [156].

The study used 16- and 64-detector MDCT scanners vs. four-detector scanners in NLST. Software allowed 3D volumetric rather than 2D linear measurements. Nodules would be classified as low, intermediate, or high risk based on size, newness, and other features. Low-risk nodules would be recorded and the patient would return for the next scheduled screening scan. High-risk nodules would be referred immediately to a pulmonologist for biopsy; if negative the patient would stay in the study for the next scheduled screening. Intermediate-sized nodules would receive a repeat CT after 1 ½ to 3 months per protocol to calculate growth by volume doubling time. Growth rates faster than specified cutoffs would be referred to a pulmonologist for further evaluation. If cancer was diagnosed, the patient would then leave the study. If negative, they would return for further screening rounds. NELSON 16-detector MDCTs emitted 2 mSv for all exams [157]. The protocol was designed to minimize repeat scanning of intermediate lesions to one CT, after which a decision to refer or not would be made.

The final results were published in The New England Journal of Medicine in 2020 [158]. After randomization of 13,195 men, 344 cancers in the screening group and 304 in controls were found. 203 of the 344 cancers in the screening group were actually detected by screening - 44 were diagnosed between screenings and 97 after cessation of screening. Almost 60% of cancers detected by screening were stage I vs. 14% in the control group. Mortality over 10 years was 24% lower in men and 33% lower in a separate analysis of women (only 2594 subjects were women, due to low smoking prevalence). Benefit was seen even with the longer 2-year intervals after the first two screenings, demonstrating that less frequent screening in those already proven "negative" twice was viable. Overdiagnosis was estimated at 10%, and the NNS to save one life was 132. They found that after the first round, approximately 20% of participants required a repeat CT within 3 months to assess intermediate nodules, similar to other studies. However, for rounds 2-4, only 1.9-6.7% required a short-term repeat CT, and over half of new nodules resolved. Only about 2% of the CTs were "positive" in each round, and about one-half of those had cancer, giving a false positive rate of about 1.2%. In NLST the false positive rate was almost 25%. All age and smoking sub-groups showed mortality benefit; examining only NLST-eligible NELSON patients, a trend towards mortality benefit exceeding NLST was seen. The NEJM editorial by Duffy stated: "With the NELSON results, the efficacy of low-dose CT screening for lung cancer is confirmed. Our job is no longer to assess whether low-dose CT screening for lung cancer works: it does. Our job is to identify the target population in which it will be acceptable and cost-effective" [159].

Future of Lung Cancer Screening

Of note, the authors of NLST published an extended 12-year update on their subjects in 2019. At 12 years there were 1701 LC cases in the LDCT arm vs. 1681 in the CXR arm, thus showing equal incidence due to catch-up. Stage shift was preserved, i.e., more stage I (39.6% vs. 27.5%) and less stage IV (27.5% vs. 35.5%), and was similar to the original.

Overdiagnosis was down to 3%. Relative reduction in LC mortality fell to 11% (previously 20%), and there was no benefit in all-cause mortality, which had been 6.7%. Likely long follow-up period in an older cohort allowed catch-up in mortality for other reasons. NNS remained stable at 303, which signified that the signal of LC deaths prevented by screening had lasted for a decade and was due to cure, not delay in death. The conclusion was that there was still benefit even more than a decade, further reinforcing the effectiveness of LCS.

The Microsimulation Screening Analysis Lung (MISCAN-Lung) model was used to project the ramifications of full-scale LCS, and it found that there would be less need for chemotherapy and radiotherapy but there would be major increase in demand for surgery above current capacity. Careful planning and increasing the number of thoracic surgeons as quickly as possible would be imperative [160].

There is likely benefit to screening patients who do not meet CMS criteria and defining this group is a current topic of debate. Some advocate criteria based on personal risk calculation [161], and others advocate selection of those who could live longer after screening [162].

Text messaging to increase colon, breast, and cervical cancer screening has had moderate success [163] and bears further study for LCS. There is currently a multi-center clinical trial randomizing LCS active smokers to three arms for smoking cessation: standard referral, digital-based education/reminders on phone, and computer and a tobacco treatment specialist/digital combination [164]. Creating similar clinical trials using electronic platforms for recruiting, educating, and reminding candidates of LCS may be viable.

The SARS-CoV-2 (COVID-19) pandemic has changed medical practice around the world and has affected LCS. In April 2020, the American College of Radiology and the American Thoracic Society jointly created expert consensus statements to delay the reassessment of incidental or LCS nodules with low/intermediate risk potential for at least 3-6 months and to delay LCS baseline exams for 3-6 months, to be re-examined as the situation unfolds. For high-risk nodules with 65-85% certainty of malignancy, PET-CT or biopsy to assess need for surgery or radiotherapy is recommended rather than proceeding directly to surgery as was recommended by prior guidelines. Only the highest-risk nodules with >85% certainty of malignancy, should proceed to empiric surgery or radiotherapy. The goal of the strategy was to decrease risk and conserving personal protective equipment (PPE) by reducing procedures [165]. Amit et al. wrote and editorial that questioned the postponement of LCS, pointing out that mortality for LC is higher than that for COVID-19 and warning of a "cancer boom" in the future. They advocate frequent reassessment of this policy, and upon reopening, using triage systems to prioritize those who might benefit the most. They also call for cross-disciplinary discussion, merging of programs and resources and collaboration to help decrease the coming backlog in a way that helps the most people for the lowest cost [166].

In these efforts, telemedicine can help to maintain social distancing while allowing shared decision-making sessions with a provider.

Health Economics

As stated previously, LC has very high costs in terms of expenditures and lost productivity [4]. Much of the cost occurs due to treatment and complications of late-stage disease. Cost-effectiveness studies are designed to convince insurers to pay for LCS by estimating the costs incurred for screening in return for quality life-years. A 2013 study found LCS was as cost-effective as cervical and colon screening and better than mammography, diabetes screening, HIV testing, cholesterol medication, and dialysis [167]. A 2019 CISNET computer modeling study compared costeffectiveness and mortality benefits of LCS in NLST-, CMS-, and USPSTF-eligible groups and found the CMS criteria gave the best combination of cost-effectiveness, quality lifeyears, and mortality benefit [168]. Applying NLST criteria would cost less but have less benefit while USPTF criteria would give the most benefit but also be the most expensive. The cost to insurers of expanding screening to patients using NCCN or NELSON criteria was not estimated.

One of the centers using NCCN criteria to expand LCS reported on profitability analysis for the institution. They actually offered LCS CT for free, regardless of insurance status. As a result of this screening, 60 new patients entered their healthcare system, and profitability above the initial overhead was achieved after 2 years [169]. They also noted that 73% of the lesions found were stage I, and the cost of treatment was \$10,000 to \$20,000. Costs associated with treatment of stage III or IV LCs are \$80,000 to \$100,000. The argument that healthcare systems can profit from LCS may speed their dissemination.

Cancer Prevention and Screening in the ED

Smoking cessation counseling in the ED has been studied since the early 2000s and is discussed in detail elsewhere in this book. EDs have long been involved in public health initiatives such as accident prevention, domestic violence screening, and vaccinations, among others. The idea of using EDs to promote cancer prevention is also not new. A 1996 study screened 116 women for breast and 644 for cervical cancer in a busy New York City ED with an underserved population. Nine patients with cancers underwent treatment [170]. In 2006, Cummings et al. in Canada reported using ED surveys about cervical cancer screening and immunizations and finding 307 women overdue for Pap smears. One hundred fourteen were contacted in follow-up and 54 (47%) reported receiving an outpatient Pap smear. Four of the 54 had significant pathology requiring further intervention [171]. Zun and Downey in Chicago questioned ED patients (mainly Black and Hispanic) about colon, prostate, cervical, and breast cancer screenings; of the patients who accepted referrals, 33% (breast) to 53% (colon) had confirmed appointments on follow-up [172]. In contrast, a 2008 Boston-based survey conducted in three academic EDs found that most women and a majority of men were compliant with recommended cervical, breast, testicular, and prostate cancer screening compared to general population, regardless of race or ethnicity. This population was 65% non-Hispanic White and had higher average income than most urban ED populations [173]. A 2017 analysis of the National Health Interview Survey found that patients who used the ED for health care were far less likely to have had recommended cancer screening than those with a usual source of care [126]. An initiative to promote public health topics (not including cancer screening) in a busy Maryland ED used interactive computer kiosks, noting that self-selection of motivated individuals could potentially lead to more efficient delivery of health education and even follow-up without sacrificing ED time or staff [174].

Emergency medicine organizations such as the American College of Emergency Physicians, the Society for Academic Emergency Medicine, and the American Academy of Emergency Medicine all have sections and/or polices on public health, and SAEM has an interest group dedicated to oncologic emergencies.

Lung Cancer Screening in the ED: What Is Our Role?

General barriers to screening include poor access to health care, lack of awareness or skepticism by primary care providers about LCS, fears of radiation exposure, concerns about cost, concerns about adequate interpretation of LDCTs, concerns about unnecessary invasive procedures, and concerns about a positive diagnosis, to name a few. In addition, racial/ethnic minorities, rural residents, and sexual minorities who are at high risk but medically disenfranchised face additional barriers to this potentially lifesaving procedure.

EDs act as the nation's safety net and are integral to our healthcare system. Public health and cancer prevention initiatives are already part of our profession's infrastructure and mission. We could help to increase screening in different ways. Most EDs use electronic health record (EHR) systems that allow patients to be "flagged" for various conditions, including public health concerns. EHRs have been shown to improve colon and breast cancer screening rates in ambulatory settings [175]. It would not be difficult to adapt such tools to ED EHR systems to identify such patients when they present. The teachable moment during a health emergency is often mentioned and has been shown to increase patient's willingness to change behavior, particularly with regard to substance abuse [176, 177]. Existing ED smoking cessation programs are another avenue to recruit eligible patients as many of them are likely at higher risk for LC.

Education targeting high-risk patients as they come to EDs for any reason is a relatively low-cost opportunity to improve screening rates. Use of modern technology such as computer/kiosks or smartphones, texts, and social media platforms may help by giving patients preliminary information about LCS without need for a dedicated person to do so.

Navigators for LCS programs performing outreach and facilitation for patients and providers have had some success facilitating screening in other settings, and liaison between EDs and local LCS existing navigators may serve to increase recruitment without dedicating ED personnel to the task. Partnering of community hospitals or freestanding EDs in remote areas with distant LCS centers may be facilitated by the use of telemedicine, as this modality becomes more familiar.

Any initiative into public health and cancer screening by EDs will certainly require coordination between local and state government authorities, philanthropic programs, public health programs, large medical centers, and community stakeholders. Whatever solutions that EDs adopt should not detract real-time resources from the EDs' primary mission.

Shared Decision-Making

Ideally, the burden of formal shared decision-making conversations and documentation that are a CMS requirement for LCS reimbursement should be undertaken by primary care physicians, pulmonologists, or oncologists in the proper setting [178], rather than the ED. Even if not formally part of a referral network, any ED can direct eligible patients to publicly available decision aids regarding LCS to help patients with decision-making (Table 12.1).

Table 12.1 Online resources for lung cancer screening

Agency for Healthcare Research and Quality: https://effectivehealthcare.ahrq.gov/decision-aids/lung-cancer-screening/home.html

- American Thoracic Society: https://www.thoracic.org/patients/patient-resources/resources/decision-aid-lcs.pdf
- Brock University Lung Cancer Risk Calculators: https://brocku.ca/lung-cancer-screening-and-risk-prediction/risk-calculators/

NCI (National Cancer Institute): https://www.cancer.gov/types/lung/patient/lung-screening-pdq Veterans Health Administration: https://www.prevention.va.gov/docs/LungCancerScreeningHandout.pdf

CDC (Centers for Disease Control and Prevention): https://www.cdc.gov/cancer/lung/basic_info/screening.htm

MD Anderson Cancer Center: https://www.mdanderson.org/publications/oncolog/house-call%2D%2Dlung-cancer-screening.h12-1591413. html

Table 12.2 Key points/pearls

Lung cancer in the USA is the costliest cancer in terms of productivity and lives lost

COPD and coronary disease often co-exist with lung cancer, and COPD increases lung cancer mortality

Never smoker's lung cancer (LC) incidence is rising, and patients tend to be women. In East Asia, women have very high incidence of never smoker lung cancer, likely due to genetic variation. Air pollution is another cause of never smoker cancer, as is radon, as well as many other environmental exposures and medical conditions

Black men have the highest incidence and mortality from lung cancer, but non-Hispanic White and Native American men also have very high risk and mortality; there is correlation with geography, poverty, low air quality, and poor access to medical care. Others at risk include LGBT, disabled individuals, and enlisted military personnel. ED diagnosis of LC is associated with poor outcome

Shortness of breath and altered mental status correlate with ICU admission and short-term death

Adenocarcinoma and squamous cell LC are curable with early detection and surgical treatment but have poor prognosis if not resectable. Small-cell lung cancer remains highly deadly, even if found early

Incidental CT masses >3 cm are likely malignant and should be referred for further diagnosis. In adults 35 and older, lesions <6 mm in diameter are very unlikely to be malignant, unless the patient has history of cancer and immunosuppression or the lesions have irregular or spiculated morphology or are in the upper lobe location. Multiple nodules (2–4) are more likely malignant, but 5 or greater are less likely (unless there is history of cancer). Radiologists use Fleischner Society Guidelines for follow-up recommendations of suspicious lesions as part of the reading

Lung cancer screening is recommended by most medical organizations and approved by CMS for active or former smokers 55–77, with 30 pack-years of smoking, and those who have quit less than 15 years ago

Another high-risk group are those age >50 with 20+ pack-years *plus* history of cancer, COPD/pulmonary fibrosis, family history of LC, radon exposure, or occupational carcinogen exposure

Only 4% of eligible patients receive lung cancer screening (LCS) despite the test's approval by Centers for Medicare and Medicaid Services (CMS)

Overdiagnosis and false positives can lead to unnecessary distress, radiation, and procedures, but decreasing unintentional harm has been a priority in LCS research

Radiation dose from LDCT is about 10 times that of CXR, and one-fifth the dose of standard chest CT. As technology and protocols advance, the dose may soon approach that of 1–2 CXRs

Emergency physicians and departments can help increase LCS by "flagging" eligible patients using electronic medical records and then working with local LCS programs approved by the American College of Radiology. Many of these are developing outreach and navigator programs to increase recruitment and may be able to provide education and logistical support to interested patients. Use of telemedicine, cell phone, and social media platforms can be integrated. Coordination with local and state government authorities, philanthropic programs, public health programs, large medical centers, and community stakeholders is essential to maximize effectiveness and minimize dilution of ED resources

Conclusion (Table 12.2)

LC is the deadliest of all cancers in the USA and across the world. Despite advances in treatment, later-stage disease remains highly fatal. Early diagnosis at an early localized stage is the best chance for cure, and mass screening that had been unsuccessfully attempted for decades is only recently possible with LDCT. The potential for saving lives and costs makes increased screening for early LC a public health priority. Increasing the numbers of eligible patients who undergo screening should be approached on many different fronts.

As the safety net for society, the ED should certainly be able to assist with recruitment and referral. Real-time use of LDCT in eligible patients where CT scanners are not otherwise available may be another avenue worth exploring. However, this would only work if there is coordination with LCS programs and treatment centers. Partnerships with local community resources, advocacy in government circles at the regional and state level, and recruitment of philanthropic support are likely keys to success.

References

- Ferlay J, Colombet M, Soerjomataram I, Mathers C, Parkin DM, Piñeros M, et al. Estimating the global cancer incidence and mortality in 2018: GLOBOCAN sources and methods. Int J Cancer. 2019;144(8):1941–53.
- Bray F, Ferlay J, Soerjomataram I, Siegel RL, Torre LA, Jemal A. Global cancer statistics 2018: GLOBOCAN estimates of incidence and mortality worldwide for 36 cancers in 185 countries. CA Cancer J Clin. 2018;68(6):394–424.
- National Cancer Institute. Surveillance, epidemiology, and end results (SEER): cancer stat facts: lung and bronchus cancer. https:// seer.cancer.gov/statfacts/html/lungb.html. Accessed 23 Apr 2020.
- National Cancer Institute. Financial burden of cancer care [Internet]. Data update as of Mar 2020. https://progressreport.cancer.gov/after/economic_burden. Accessed 24 May 2020.
- Schwartz AG, Cote ML. Epidemiology of lung cancer. In: Ahmad A, Gadgeel S, editors. Lung cancer and personalized medicine. Adv Exp Med Biol, vol. 893. Cham: Springer; 2016. p. 22–41. https:// doi.org/10.1007/978-3-319-24223-1_2.
- Torre LA, Siegel RL, Jemal A. Lung cancer statistics. Adv Exp Med Biol. 2016;893:1–19.
- Barta JA, Powell CA, Wisnivesky JP. Global epidemiology of lung cancer. Ann Glob Health. 2019;85(1):8.

- Gao YH, Guan WJ, Liu Q, Wang HQ, Zhu YN, Chen RC, et al. Impact of COPD and emphysema on survival of patients with lung cancer: a meta-analysis of observational studies. Respirology. 2016;21(2):269–79.
- Husebo GR, Nielsen R, Hardie J, Bakke PS, Lerner L, D'Alessandro-Gabazza C, et al. Risk factors for lung cancer in COPD – results from the Bergen COPD cohort study. Respir Med. 2019;152:81–8.
- Zhang LR, Morgenstern H, Greenland S, Chang S-C, Lazarus P, Teare MD, et al. Cannabis smoking and lung cancer risk: pooled analysis in the International Lung Cancer Consortium. Int J Cancer. 2015;136(4):894–903.
- de Groot P, Munden RF. Lung cancer epidemiology, risk factors, and prevention. Radiol Clin N Am. 2012;50(5):863–76.
- Pallis AG, Syrigos KN. Lung cancer in never smokers: disease characteristics and risk factors. Crit Rev Oncol Hematol. 2013;88(3):494–503.
- Pelosof L, Ahn C, Gao A, Horn L, Madrigales A, Cox J, et al. Proportion of never-smoker non-small cell lung cancer patients at three diverse institutions. J Natl Cancer Inst. 2017;109(7):djw295.
- Sun S, Schiller JH, Gazdar AF. Lung cancer in never smokers—a different disease. Nat Rev Cancer. 2007;7(10):778–90.
- Wakelee HA, Chang ET, Gomez SL, Keegan TH, Feskanich D, Clarke CA, et al. Lung cancer incidence in never smokers. J Clin Oncol. 2007;25(5):472–8.
- Corrales L, Rosell R, Cardona AF, Martin C, Zatarain-Barron ZL, Arrieta O. Lung cancer in never smokers: the role of different risk factors other than tobacco smoking. Crit Rev Oncol Hematol. 2020;148:102895.
- Krewski D, Lubin JH, Zielinski JM, Alavanja M, Catalan VS, Field RW, et al. Residential radon and risk of lung cancer: a combined analysis of 7 North American case-control studies. Epidemiology. 2005;16(2):137–45.
- Vineis P, Forastiere F, Hoek G, Lipsett M. Outdoor air pollution and lung cancer: recent epidemiologic evidence. Int J Cancer. 2004;111(5):647–52.
- Raaschou-Nielsen O, Andersen ZJ, Hvidberg M, Jensen SS, Ketzel M, Sørensen M, et al. Lung cancer incidence and long-term exposure to air pollution from traffic. Environ Health Perspect. 2011;119(6):860–5.
- Turner MC, Krewski D, Pope CA 3rd, Chen Y, Gapstur SM, Thun MJ. Long-term ambient fine particulate matter air pollution and lung cancer in a large cohort of never-smokers. Am J Respir Crit Care Med. 2011;184(12):1374–81.
- Underwood JM, Townsend JS, Tai E, Davis SP, Stewart SL, White A, et al. Racial and regional disparities in lung cancer incidence. Cancer. 2012;118(7):1910–8.
- 22. Hosgood HD, Boffetta P, Greenland S, Lee Y-CA, McLaughlin J, Seow A, et al. In-home coal and wood use and lung cancer risk: a pooled analysis of the International Lung Cancer Consortium. Environ Health Perspect. 2010;118(12):1743–7.
- Dresler C. The changing epidemic of lung cancer and occupational and environmental risk factors. Thorac Surg Clin. 2013;23(2):113–22.
- Pampel F. Tobacco use in sub-Sahara Africa: estimates from the demographic health surveys. Soc Sci Med. 2008;66(8):1772–83.
- Jemal A, Bray F, Forman D, O'Brien M, Ferlay J, Center M, et al. Cancer burden in Africa and opportunities for prevention. Cancer. 2012;118(18):4372–84.
- Torre LA, Siegel RL, Ward EM, Jemal A. Global cancer incidence and mortality rates and trends--an update. Cancer Epidemiol Biomark Prev. 2016;25(1):16–27.
- 27. Goss PE, Strasser-Weippl K, Lee-Bychkovsky BL, Fan L, Li J, Chavarri-Guerra Y, et al. Challenges to effective cancer control in China, India, and Russia. Lancet Oncol. 2014;15(5):489–538.

- Torre LA, Islami F, Siegel RL, Ward EM, Jemal A. Global cancer in women: burden and trends. Cancer Epidemiol Biomark Prev. 2017;26(4):444–57.
- Torre LA, Siegel RL, Ward EM, Jemal A. International variation in lung cancer mortality rates and trends among women. Cancer Epidemiol Biomark Prev. 2014;23(6):1025–36.
- Nelson DE, Mowery P, Asman K, Pederson LL, O'Malley PM, Malarcher A, et al. Long-term trends in adolescent and young adult smoking in the United States: metapatterns and implications. Am J Public Health. 2008;98(5):905–15.
- Siegel RL, Miller KD, Jemal A. Cancer statistics, 2019. CA Cancer J Clin. 2019;69(1):7–34.
- 32. Singh GK, Jemal A. Socioeconomic and racial/ethnic disparities in cancer mortality, incidence, and survival in the United States, 1950–2014: over six decades of changing patterns and widening inequalities. J Environ Public Health. 2017;2017:2819372.
- 33. Atkins GT, Kim T, Munson J. Residence in rural areas of the United States and lung cancer mortality: disease incidence, treatment disparities, and stage-specific survival. Ann Am Thorac Soc. 2017;14(3):403–11.
- Klugman M, Xue X, Hosgood HD. Race/ethnicity and lung cancer survival in the United States: a meta-analysis. Cancer Causes Control. 2019;30(11):1231–41.
- 35. Drope J, Liber AC, Cahn Z, Stoklosa M, Kennedy R, Douglas CE, et al. Who's still smoking? Disparities in adult cigarette smoking prevalence in the United States. CA: a cancer journal for clinicians. 2018;68(2):106–15.
- Kalemkerian GP, Akerley W, Bogner P, Borghaei H, Chow LQ, Downey RJ, et al. Small cell lung cancer. J Natl Compr Cancer Netw. 2013;11(1):78–98.
- Lewis DR, Check DP, Caporaso NE, Travis WD, Devesa SS. US lung cancer trends by histologic type. Cancer. 2014;120(18):2883–92.
- Osmani L, Askin F, Gabrielson E, Li QK. Current WHO guidelines and the critical role of immunohistochemical markers in the subclassification of non-small cell lung carcinoma (NSCLC): moving from targeted therapy to immunotherapy. Semin Cancer Biol. 2018;52(Pt 1):103–9.
- Zheng M. Classification and pathology of lung cancer. Surg Oncol Clin North Am. 2016;25(3):447–68.
- 40. Travis WD, Brambilla E, Nicholson AG, Yatabe Y, Austin JHM, Beasley MB, et al. The 2015 World Health Organization Classification of Lung Tumors: impact of genetic, clinical and radiologic advances since the 2004 classification. J Thorac Oncol. 2015;10(9):1243–60.
- 41. Kalemkerian GP, Schneider BJ. Advances in small cell lung cancer. Hematol Oncol Clin North Am. 2017;31(1):143–56.
- Pacheco J, Bunn PA. Advancements in small-cell lung cancer: the changing landscape following IMpower-133. Clin Lung Cancer. 2019;20(3):148–60, e2.
- Flehinger BJ, Kimmel M, Melamed MR. The effect of surgical treatment on survival from early lung cancer. Implications for screening. Chest. 1992;101(4):1013–8.
- Melamed MR, Flehinger BJ, Zaman MB. Impact of early detection on the clinical course of lung cancer. Surg Clin North Am. 1987;67(5):909–24.
- Nesbitt JC, Putnam JB Jr, Walsh GL, Roth JA, Mountain CF. Survival in early-stage non-small cell lung cancer. Ann Thorac Surg. 1995;60(2):466–72.
- 46. Strand TE, Rostad H, Møller B, Norstein J. Survival after resection for primary lung cancer: a population based study of 3211 resected patients. Thorax. 2006;61(8):710–5.
- 47. Chang MY, Mentzer SJ, Colson YL, Linden PA, Jaklitsch MT, Lipsitz SR, et al. Factors predicting poor survival after resection of stage IA non-small cell lung cancer. J Thorac Cardiovasc Surg. 2007;134(4):850–6.

- Medford-Davis L, Decamp M, Recht A, Flickinger J, Belani CP, Varlotto J. Surgical management of early-stage non-small cell lung carcinoma and the present and future roles of adjuvant therapy: a review for the radiation oncologist. Int J Radiat Oncol Biol Phys. 2012;84(5):1048–57.
- 49. Osarogiagbon RU, Veronesi G, Fang W, Ekman S, Suda K, Aerts JG, et al. Early-stage NSCLC: advances in thoracic oncology 2018. J Thorac Oncol. 2019;14(6):968–78.
- Forde PM, Chaft JE, Smith KN, Anagnostou V, Cottrell TR, Hellmann MD, et al. Neoadjuvant PD-1 blockade in resectable lung cancer. N Engl J Med. 2018;378(21):1976–86.
- Bradley CJ, Yabroff KR, Mariotto AB, Zeruto C, Tran Q, Warren JL. Antineoplastic treatment of advanced-stage non-small-cell lung cancer: treatment, survival, and spending (2000 to 2011). J Clin Oncol. 2017;35(5):529–35.
- Bagcchi S. Lung cancer survival only increases by a small amount despite recent treatment advances. Lancet Respir Med. 2017;5(3):169.
- 53. Rivera DR, Gallicchio L, Brown J, Liu B, Kyriacou DN, Shelburne N. Trends in adult cancer-related emergency department utilization: an analysis of data from the Nationwide Emergency Department Sample. JAMA Oncol. 2017;3(10):e172450.
- 54. Caterino JM, Adler D, Durham DD, Yeung SJ, Hudson MF, Bastani A, et al. Analysis of diagnoses, symptoms, medications, and admissions among patients with cancer presenting to emergency departments. JAMA Netw Open. 2019;2(3):e190979.
- 55. Elsayem AF, Merriman KW, Gonzalez CE, Yeung SC, Chaftari PS, Reyes-Gibby C, et al. Presenting Symptoms in the emergency department as predictors of intensive care unit admissions and hospital mortality in a comprehensive cancer center. J Oncol Pract. 2016;12(5):e554–63.
- Zhou Y, Abel GA, Hamilton W, Pritchard-Jones K, Gross CP, Walter FM, et al. Diagnosis of cancer as an emergency: a critical review of current evidence. Nat Rev Clin Oncol. 2017;14(1):45–56.
- Braithwaite S, Perina D. Dyspnea. In: Walls RM, Hockberger RS, Gausche-Hill M, editors. Rosen's emergency medicine: concepts and clinical practice. 9th ed. Philadelphia: Elsevier Health Sciences; 2018. p. 195–203.
- Kocher F, Hilbe W, Seeber A, Pircher A, Schmid T, Greil R, et al. Longitudinal analysis of 2293 NSCLC patients: a comprehensive study from the TYROL registry. Lung Cancer. 2015;87(2):193–200.
- Hyde L, Hyde CI. Clinical manifestations of lung cancer. Chest. 1974;65(3):299–306.
- 60. Chernow B, Sahn SA. Carcinomatous involvement of the pleura: an analysis of 96 patients. Am J Med. 1977;63(5):695–702.
- Patel AM, Peters SG. Clinical manifestations of lung cancer. Mayo Clin Proc. 1993;68(3):273–7.
- 62. Hamilton W, Peters TJ, Round A, Sharp D. What are the clinical features of lung cancer before the diagnosis is made? A population based case-control study. Thorax. 2005;60(12):1059–65.
- Shim J, Brindle L, Simon M, George S. A systematic review of symptomatic diagnosis of lung cancer. Fam Pract. 2014;31(2):137–48.
- 64. Okoli GN, Kostopoulou O, Delaney BC. Is symptom-based diagnosis of lung cancer possible? A systematic review and meta-analysis of symptomatic lung cancer prior to diagnosis for comparison with real-time data from routine general practice. PLoS One. 2018;13(11):e0207686.
- Sikka V, Ornato JP. Cancer diagnosis and outcomes in Michigan EDs vs other settings. Am J Emerg Med. 2012;30(2):283–92.
- 66. Heron SL, Stettner E, Haley LL Jr. Racial and ethnic disparities in the emergency department: a public health perspective. Emerg Med Clin North Am. 2006;24(4):905–23.
- 67. Clinton JE, Yaron M, Tsai SH. Chest radiography in the emergency department. Ann Emerg Med. 1986;15(3):254–6.
- Butcher BL, Nichol KL, Parenti CM. High yield of chest radiography in walk-in clinic patients with chest symptoms. J Gen Intern Med. 1993;8(3):115–9.

- van Rossum AB, Pattynama PM, Ton ER, Treurniet FE, Arndt JW, van Eck B, et al. Pulmonary embolism: validation of spiral CT angiography in 149 patients. Radiology. 1996;201(2):467–70.
- Patel S, Kazerooni EA. Helical CT for the evaluation of acute pulmonary embolism. AJR Am J Roentgenol. 2005;185(1):135–49.
- Manghat NE, Morgan-Hughes GJ, Roobottom CA. Multi-detector row computed tomography: imaging in acute aortic syndrome. Clin Radiol. 2005;60(12):1256–67.
- 72. Rubinshtein R, Halon DA, Gaspar T, Jaffe R, Karkabi B, Flugelman MY, et al. Usefulness of 64-slice cardiac computed tomographic angiography for diagnosing acute coronary syndromes and predicting clinical outcome in emergency department patients with chest pain of uncertain origin. Circulation. 2007;115(13):1762–8.
- Park JE, Kim Y, Lee SW, Shim SS, Lee JK, Lee JH. The usefulness of low-dose CT scan in elderly patients with suspected acute lower respiratory infection in the emergency room. Br J Radiol. 2016;89(1060):20150654.
- 74. Gezer NS, Balci P, Tuna KC, Akin IB, Baris MM, Oray NC. Utility of chest CT after a chest X-ray in patients presenting to the ED with non-traumatic thoracic emergencies. Am J Emerg Med. 2017;35(4):623–7.
- 75. Richman PB, Courtney DM, Friese J, Matthews J, Field A, Petri R, et al. Prevalence and significance of nonthromboembolic findings on chest computed tomography angiography performed to rule out pulmonary embolism: a multicenter study of 1,025 emergency department patients. Acad Emerg Med. 2004;11(6):642–7.
- 76. Kino A, Boiselle PM, Raptopoulos V, Hatabu H. Lung cancer detected in patients presenting to the Emergency Department studies for suspected pulmonary embolism on computed tomography pulmonary angiography. Eur J Radiol. 2006;58(1):119–23.
- Goldman IA, Cunqueiro A, Scheinfeld MH. New CT-based diagnoses of torso cancer is low in the emergency department setting. Emerg Radiol. 2019;26(6):633–8.
- Hall WB, Truitt SG, Scheunemann LP, Shah SA, Rivera MP, Parker LA, et al. The prevalence of clinically relevant incidental findings on chest computed tomographic angiograms ordered to diagnose pulmonary embolism. Arch Intern Med. 2009;169(21):1961–5.
- Lehman SJ, Abbara S, Cury RC, Nagurney JT, Hsu J, Goela A, et al. Significance of cardiac computed tomography incidental findings in acute chest pain. Am J Med. 2009;122(6):543–9.
- Hansell DM, Bankier AA, MacMahon H, McLoud TC, Muller NL, Remy J. Fleischner Society: glossary of terms for thoracic imaging. Radiology. 2008;246(3):697–722.
- Herzog C, Burgener FA. Lungs. In: Burgener FA, Meyers SP, Herzog C, Zaunbauer W, editors. Differential Diagnosis in Computed Tomography. 2nd edition. Stuttgart: Thieme; 2012. p. 592–629.
- Guss DA, Almazroua F. 385 impact of preliminary radiology interpretations of imaging studies in an emergency department setting. Ann Emerg Med. 2012;60(4):S135–S6.
- Tranovich MJ, Gooch C, Dougherty J. 142 overnight radiology interpretation discrepancies between emergency physicians and radiologists in a community hospital emergency department (abstract). Ann Emerg Med. 2018;72(4 Suppl):S59.
- MacMahon H, Naidich DP, Goo JM, Lee KS, Leung ANC, Mayo JR, et al. Guidelines for management of incidental pulmonary nodules detected on CT images: from the Fleischner Society 2017. Radiology. 2017;284(1):228–43.
- 85. Scholtz JE, Lu MT, Hedgire S, Meyersohn NM, Oliveira GR, Prabhakar AM, et al. Incidental pulmonary nodules in emergent coronary CT angiography for suspected acute coronary syndrome: impact of revised 2017 Fleischner Society Guidelines. J Cardiovasc Comput Tomogr. 2018;12(1):28–33.
- 86. Jha P, Ramasundarahettige C, Landsman V, Rostron B, Thun M, Anderson RN, et al. 21st-century hazards of smoking

and benefits of cessation in the United States. N Engl J Med. $2013;368(4){:}341{-}50.$

- Bowe B, Xie Y, Yan Y, Al-Aly Z. Burden of cause-specific mortality associated with PM2.5 air pollution in the United States. JAMA Netw Open. 2019;2(11):e1915834.
- Kim S, Brewster MS, Schwartz GG. Communicating radon risk via a smartphone app: a pilot intervention study. BMC Public Health. 2020;20(1):547.
- Chiavacci SJ, Shapiro CD, Pindilli EJ, Casey CF, Rayens MK, Wiggins AT, et al. Economic valuation of health benefits from using geologic data to communicate radon risk potential. Environ Health. 2020;19(1):36.
- Fontana RS, Sanderson DR, Woolner LB, Taylor WF, Miller WE, Muhm JR. Lung cancer screening: the Mayo program. J Occup Med. 1986;28(8):746–50.
- 91. Tockman MS. Survival and mortality from lung cancer in a screened population. Chest. 1986;89(4):324S–5S.
- Cho H, Mariotto AB, Schwartz LM, Luo J, Woloshin S. When do changes in cancer survival mean progress? The insight from population incidence and mortality. J Natl Cancer Inst Monogr. 2014;2014(49):187–97.
- Eddy DM. Screening for lung cancer. Ann Intern Med. 1989;111(3):232–7.
- Patz EF Jr, Goodman PC, Bepler G. Screening for lung cancer. N Engl J Med. 2000;343(22):1627–33.
- Bach PB, Kelley MJ, Tate RC, McCrory DC. Screening for lung cancer: a review of the current literature. Chest. 2003;123(1 Suppl):72S–82S.
- 96. Finigan JH, Kern JA. Lung cancer screening: past, present and future. Clin Ches Med. 2013;34(3):365–71.
- Marcus PM, Bergstralh EJ, Zweig MH, Harris A, Offord KP, Fontana RS. Extended lung cancer incidence follow-up in the Mayo Lung Project and overdiagnosis. J Natl Cancer Inst. 2006;98(11):748–56.
- Welch HG, Black WC. Overdiagnosis in cancer. J Natl Cancer Inst. 2010;102(9):605–13.
- McFarlane MJ, Feinstein AR, Wells CK. Clinical features of lung cancers discovered as a postmortem "surprise". Chest. 1986;90(4):520–3.
- 100. Behera M, Owonikoko TK, Gal AA, Steuer CE, Kim S, Pillai RN, et al. Lung adenocarcinoma staging using the 2011 IASLC/ ATS/ERS Classification: a pooled analysis of adenocarcinoma in situ and minimally invasive adenocarcinoma. Clin Lung Cancer. 2016;17(5):e57–64.
- Srivastava S, Reid BJ, Ghosh S, Kramer BS. Research needs for understanding the biology of overdiagnosis in cancer screening. J Cell Physiol. 2016;231(9):1870–5.
- 102. Patz EF Jr, Pinsky P, Gatsonis C, Sicks JD, Kramer BS, Tammemagi MC, et al. Overdiagnosis in low-dose computed tomography screening for lung cancer. JAMA Intern Med. 2014;174(2):269–74.
- 103. Thalanayar PM, Altintas N, Weissfeld JL, Fuhrman CR, Wilson DO. Indolent, potentially inconsequential lung cancers in the Pittsburgh lung screening study. Ann Am Thorac Soc. 2015;12(8):1193–6.
- 104. Oken MM, Hocking WG, Kvale PA, Andriole GL, Buys SS, Church TR, et al. Screening by chest radiograph and lung cancer mortality: the Prostate, Lung, Colorectal, and Ovarian (PLCO) randomized trial. JAMA. 2011;306(17):1865–73.
- 105. Sone S, Takashima S, Li F, Yang Z, Honda T, Maruyama Y, et al. Mass screening for lung cancer with mobile spiral computed tomography scanner. Lancet. 1998;351(9111):1242–5.
- Henschke CI, McCauley DI, Yankelevitz DF, Naidich DP, McGuinness G, Miettinen OS, et al. Early Lung Cancer Action

Project: overall design and findings from baseline screening. Lancet. 1999;354(9173):99–105.

- 107. Naidich DP, Marshall CH, Gribbin C, Arams RS, McCauley DI. Low-dose CT of the lungs: preliminary observations. Radiology. 1990;175(3):729–31.
- 108. Kaneko M, Eguchi K, Ohmatsu H, Kakinuma R, Naruke T, Suemasu K, et al. Peripheral lung cancer: screening and detection with low-dose spiral CT versus radiography. Radiology. 1996;201(3):798–802.
- 109. Paci E. Lung cancer screening: the methodological debate. Lung Cancer. 2002;38(1):S17–21.
- 110. Henschke CI, Yankelevitz DF, Libby DM, Pasmantier MW, Smith JP, Miettinen OS. Survival of patients with stage I lung cancer detected on CT screening. N Engl J Med. 2006;355(17):1763–71.
- 111. Paci E. Observational, one-arm studies and randomized population-based trials for evaluation of the efficacy of lung cancer screening. J Thorac Oncol. 2007;2(5 Suppl 1):S45–S6.
- 112. Welch HG, Woloshin S, Schwartz LM, Gordis L, Gøtzsche PC, Harris R, et al. Overstating the evidence for lung cancer screening: the International Early Lung Cancer Action Program (I-ELCAP) study. Arch Intern Med. 2007;167(21):2289–95.
- 113. National Lung Screening Trial Research Team, Aberle DR, Adams AM, Berg CD, Black WC, Clapp JD, et al. Reduced lung-cancer mortality with low-dose computed tomographic screening. N Engl J Med. 2011;365(5):395–409.
- Silvestri GA. Screening for lung cancer: it works, but does it really work? Ann Intern Med. 2011;155(8):537–9.
- 115. Bach PB, Mirkin JN, Oliver TK, Azzoli CG, Berry DA, Brawley OW, et al. Benefits and harms of CT screening for lung cancer: a systematic review. JAMA. 2012;307(22):2418–29.
- 116. American Academy of Family Physicians. Clinical preventive service recommendation: lung cancer 2013. https://www.aafp. org/patient-care/clinical-recommendations/all/lung-cancer.html. Accessed 28 Jun 2020.
- 117. Humphrey LL, Deffebach M, Pappas M, Baumann C, Artis K, Mitchell JP, et al. Screening for lung cancer with low-dose computed tomography: a systematic review to update the US Preventive services task force recommendation. Ann Intern Med. 2013;159(6):411–20.
- 118. Meza R, ten Haaf K, Kong CY, Erdogan A, Black WC, Tammemagi MC, et al. Comparative analysis of 5 lung cancer natural history and screening models that reproduce outcomes of the NLST and PLCO trials. Cancer. 2014;120(11):1713–24.
- 119. Moyer VA, Force USPST. Screening for lung cancer: U.S. Preventive Services Task Force recommendation statement. Ann Intern Med. 2014;160(5):330-8.
- 120. CMS.gov. Decision memo for screening for lung cancer with low dose computed tomography (LCDT) (CAG-00439N). Baltimore: Centers for Medicare & Medicaid Services; 2015. https://www. cms.gov/medicare-coverage-database/details/nca-decisionmemo.aspx?NCAId=274. Accessed 28 Jun 2020.
- 121. Fintelmann FJ, Bernheim A, Digumarthy SR, Lennes IT, Kalra MK, Gilman MD, et al. The 10 pillars of lung cancer screening: rationale and logistics of a lung cancer screening program. Radiographics. 2015;35(7):1893–908.
- 122. American College of Radiology. Lung cancer screening registry. Last updated 19 Mar 2018.https://www.acr.org/ Practice-Management-Quality-Informatics/Registries/Lung-Cancer-Screening-Registry. Accessed 29 Jun 2020.
- 123. American College of Radiology. Lung CT screening reporting & data system (Lung-RADS) 2014. https://www.acr.org/Clinical-Resources/Reporting-and-Data-Systems/Lung-Rads. Accessed 20 Jun 2020.

- 124. Pinsky PF, Gierada DS, Black W, Munden R, Nath H, Aberle D, et al. Performance of Lung-RADS in the National Lung Screening Trial: a retrospective assessment. Ann Intern Med. 2015;162(7):485–91.
- 125. Jemal A, Fedewa SA. Lung cancer screening with low-dose computed tomography in the United States – 2010 to 2015. JAMA Oncol. 2017;3(9):1278–81.
- 126. White A, Thompson TD, White MC, Sabatino SA, de Moor J, Doria-Rose PV, et al. Cancer screening test use – United States, 2015. MMWR Morb Mortal Wkly Rep. 2017;66(8):201–6.
- 127. Wood DE, Eapen GA, Ettinger DS, Hou L, Jackman D, Kazerooni E, et al. Lung cancer screening. J Natl Compr Cancer Netw. 2012;10(2):240–65.
- Wood DE. National Comprehensive Cancer Network (NCCN) clinical practice guidelines for lung cancer screening. Thorac Surg Clin. 2015;25(2):185–97.
- 129. Pinsky PF, Berg CD. Applying the National Lung Screening Trial eligibility criteria to the US population: what percent of the population and of incident lung cancers would be covered? J Med Screen. 2012;19(3):154–6.
- Pinsky PF, Kramer BS. Lung cancer risk and demographic characteristics of current 20-29 pack-year smokers: implications for screening. J Natl Cancer Inst. 2015;107(11):djv226.
- 131. Tammemagi MC, Church TR, Hocking WG, Silvestri GA, Kvale PA, Riley TL, et al. Evaluation of the lung cancer risks at which to screen ever- and never-smokers: screening rules applied to the PLCO and NLST cohorts. PLoS Med. 2014;11(12):e1001764.
- 132. National Comprehensive Cancer Network. Selection of individuals for lung screening. 2019. In: NCCN clinical practice guidelines in oncology lung cancer screening version 12020–14 May 2019, MS-8-MS-11. https://www.nccn.org/professionals/physician_gls/pdf/lung_screening.pdf. Accessed 14 Jul 2020.
- 133. Simmerman EL, Thomson NB 3rd, Dillard TA, Hao Z, Sadek RF, Khleif SN, et al. Free lung cancer screening trends toward a twofold increase in lung cancer prevalence in the underserved southeastern United States. South Med J. 2017;110(3):188–94.
- 134. McKee BJ, Regis S, Borondy-Kitts AK, Hashim JA, French RJ, Wald C, et al. NCCN guidelines as a model of extended criteria for lung cancer screening. J Natl Compr Cancer Netw. 2018;16(4):444–9.
- 135. Jaklitsch MT, Jacobson FL, Austin JH, Field JK, Jett JR, Keshavjee S, et al. The American Association for Thoracic Surgery guidelines for lung cancer screening using low-dose computed tomography scans for lung cancer survivors and other high-risk groups. J Thorac Cardiovasc Surg. 2012;144(1):33–8.
- Lei F, Lee E. Barriers to lung cancer screening with low-dose computed tomography. Oncol Nurs Forum. 2019;46(2):E60–71.
- 137. Wang GX, Baggett TP, Pandharipande PV, Park ER, Percac-Lima S, Shepard JO, et al. Barriers to lung cancer screening engagement from the patient and provider perspective. Radiology. 2019;290(2):278–87.
- Borondy Kitts AK. The patient perspective on lung cancer screening and health disparities. J Am Coll Radiol. 2019;16(4 Pt B):601–6.
- 139. Annangi S, Nutalapati S, Foreman MG, Pillai R, Flenaugh EL. Potential racial disparities using current lung cancer screening guidelines. J Racial Ethn Health Disparities. 2019;6(1):22–6.
- 140. Balekian AA, Wisnivesky JP, Gould MK. Surgical disparities among patients with stage i lung cancer in the national lung screening trial. Chest. 2019;155(1):44–52.
- 141. Odahowski CL, Zahnd WE, Eberth JM. Challenges and opportunities for lung cancer screening in Rural America. J Am Coll Radiol. 2019;16(4 Pt B):590–5.
- 142. Eberth JM, Bozorgi P, Lebron LM, Bills SE, Hazlett LJ, Carlos RC, et al. Geographic availability of low-dose computed tomog-

raphy for lung cancer screening in the United States, 2017. Prev Chronic Dis. 2018;15:E119.

- Triplette M, Thayer JH, Pipavath SN, Crothers K. Poor uptake of lung cancer screening: opportunities for improvement. J Am Coll Radiol. 2019;16(4 Pt A):446–50.
- 144. Huo J, Shen C, Volk RJ, Y-CT S. Use of CT and chest radiography for lung cancer screening before and after publication of screening guidelines: intended and unintended uptake. JAMA Intern Med. 2017;177(3):439–41.
- 145. Veliz P, Matthews AK, Arslanian-Engoren C, Evans-Polce RJ, Lee JGL, Boyd CJ, et al. LDCT lung cancer screening eligibility and use of CT scans for lung cancer among sexual minorities. Cancer Epidemiol. 2019;60:51–4.
- 146. Fujiwara M, Inagaki M, Nakaya N, Fujimori M, Higuchi Y, Hayashibara C, et al. Cancer screening participation in schizophrenic outpatients and the influence of their functional disability on the screening rate: a cross-sectional study in Japan. Psych Clin Neurosci. 2017;71(12):813–25.
- 147. Larke FJ, Kruger RL, Cagnon CH, Flynn MJ, McNitt-Gray MM, Wu X, et al. Estimated radiation dose associated with low-dose chest CT of average-size participants in the National Lung Screening Trial. AJR Am J Roentgenol. 2011;197(5):1165–9.
- 148. Centers for Disease Control and Prevention. Radiation and health. Radiation in your life Last reviewed 7 Dec 2015. https://www.cdc. gov/nceh/radiation/sources.html#objects. Accessed 2 Jul 2020.
- 149. Smith-Bindman R, Lipson J, Marcus R, Kim K-P, Mahesh M, Gould R, et al. Radiation dose associated with common computed tomography examinations and the associated lifetime attributable risk of cancer. Arch Intern Med. 2009;169(22):2078–86.
- McCunney RJ, Li J. Radiation risks in lung cancer screening programs: a comparison with nuclear industry workers and atomic bomb survivors. Chest. 2014;145(3):618–24.
- 151. Rampinelli C, De Marco P, Origgi D, Maisonneuve P, Casiraghi M, Veronesi G, et al. Exposure to low dose computed tomography for lung cancer screening and risk of cancer: secondary analysis of trial data and risk-benefit analysis. BMJ. 2017;356:j347.
- 152. Cattaneo SM 2nd, Meisenberg BR, Geronimo MCM, Bhandari B, Maxted JW, Brady-Copertino CJ. Lung cancer screening in the community setting. Ann Thorac Surg. 2018;105(6):1627–32.
- 153. Handy JR Jr, Skokan M, Rauch E, Zinck S, Sanborn RE, Kotova S, et al. Results of lung cancer screening in the community. Ann Fam Med. 2020;18(3):243–9.
- 154. Miller AR, Jackson D, Hui C, Deshpande S, Kuo E, Hamilton GS, et al. Lung nodules are reliably detectable on ultra-low-dose CT utilising model-based iterative reconstruction with radiation equivalent to plain radiography. Clin Radiol. 2019;74(5):409, e17–e22.
- 155. Xu DM, Gietema H, de Koning H, Vernhout R, Nackaerts K, Prokop M, et al. Nodule management protocol of the NELSON randomised lung cancer screening trial. Lung Cancer. 2006;54(2):177–84.
- 156. van Iersel CA, de Koning HJ, Draisma G, Mali WP, Scholten ET, Nackaerts K, et al. Risk-based selection from the general population in a screening trial: selection criteria, recruitment and power for the Dutch-Belgian randomised lung cancer multi-slice CT screening trial (NELSON). Int J Cancer. 2007;120(4):868–74.
- 157. Rampinelli C, Origgi D, Bellomi M. Low-dose CT: technique, reading methods and image interpretation. Cancer Imaging. 2013;12(3):548–56.
- 158. de Koning HJ, van der Aalst CM, de Jong PA, Scholten ET, Nackaerts K, Heuvelmans MA, et al. Reduced lung-cancer mortality with volume CT screening in a randomized trial. N Engl J Med. 2020;382(6):503–13.

- Duffy SW, Field JK. Mortality reduction with low-dose CT screening for lung cancer. N Engl J Med. 2020;382(6):572–3.
- 160. Blom EF, Ten Haaf K, Arenberg DA, de Koning HJ. Treatment capacity required for full-scale implementation of lung cancer screening in the United States. Cancer. 2019;125(12):2039–48.
- 161. Katki HA, Kovalchik SA, Berg CD, Cheung LC, Chaturvedi AK. Development and validation of risk models to select ever-smokers for CT lung cancer screening. JAMA. 2016;315(21):2300–11.
- 162. Cheung LC, Berg CD, Castle PE, Katki HA, Chaturvedi AK. Lifegained-based versus risk-based selection of smokers for lung cancer screening. Ann Intern Med. 2019;171(9):623–32.
- 163. Uy C, Lopez J, Trinh-Shevrin C, Kwon SC, Sherman SE, Liang PS. Text messaging interventions on cancer screening rates: a systematic review. J Med Internet Res. 2017;19(8):e296.
- 164. Graham AL, Burke MV, Jacobs MA, Cha S, Croghan IT, Schroeder DR, et al. An integrated digital/clinical approach to smoking cessation in lung cancer screening: study protocol for a randomized controlled trial. Trials. 2017;18(1):568.
- 165. Mazzone PJ, Gould MK, Arenberg DA, Chen AC, Choi HK, Detterbeck FC, et al. Management of Lung Nodules and Lung Cancer Screening During the COVID-19 Pandemic: CHEST Expert Panel Report. Chest. 2020;158(1):406–15.
- 166. Amit M, Tam S, Bader T, Sorkin A, Benov A. Pausing cancer screening during the severe acute respiratory syndrome coronavirus 2pandemic: Should we revisit the recommendations?. European Journal of Cancer. 2020;134:86–89.
- 167. Villanti AC, Jiang Y, Abrams DB, Pyenson BS. A cost-utility analysis of lung cancer screening and the additional benefits of incorporating smoking cessation interventions. PLoS One. 2013;8(8):e71379.
- 168. Criss SD, Cao P, Bastani M, Ten Haaf K, Chen Y, Sheehan DF, et al. Cost-effectiveness analysis of lung cancer screening in the United States: a comparative modeling study. Ann Intern Med. 2019;171(11):796–804.

- 169. Chung JM, Simmerman EL, Sadek RF, Wojtowicz S, Dillard TA, Albo D, et al. Financial analysis of free lung cancer screening program shows profitability using broader NCCN guidelines. Ann Thorac Surg. 2019;107(3):885–90.
- 170. Mandelblatt J, Freeman H, Winczewski D, Cagney K, Williams S, Trowers R, et al. Implementation of a breast and cervical cancer screening program in a public hospital emergency department. Cancer Control Center of Harlem. Ann Emerg Med. 1996;28(5):493–8.
- 171. Cummings GE, Francescutti LH, Predy G, Cummings G. Health promotion and disease prevention in the emergency department: a feasibility study. CJEM. 2006;8(2):100–5.
- 172. Zun LS, Downey L. Adult health screening and referral in the emergency department. South Med J. 2006;99(9):940–8.
- 173. Ginde AA, Millen JC, Love JS, Pang JM, Camargo CA Jr. Compliance with recommended cancer screening among emergency department patients: a multicenter survey. Acad Emerg Med. 2008;15(5):483–6.
- 174. Orlando MS, Rothman RE, Woodfield A, Gauvey-Kern M, Peterson S, Miller T, et al. Public health information delivery in the emergency department: analysis of a kiosk-based program. J Emerg Med. 2016;50(2):223–7.
- 175. Kern LM, Barrón Y, Dhopeshwarkar RV, Edwards A, Kaushal R, HITEC Investigators. Electronic health records and ambulatory quality of care. J Gen Intern Med. 2013;28(4):496–503.
- 176. Woodruff SI, Eisenberg K, McCabe CT, Clapp JD, Hohman M. Evaluation of California's alcohol and drug screening and brief intervention project for emergency department patients. West J Emerg Med. 2013;14(3):263–70.
- 177. Lemhoefer C, Rabe GL, Wellmann J, Bernstein SL, Cheung KW, McCarthy WJ, et al. Emergency department-initiated tobacco control: update of a systematic review and meta-analysis of randomized controlled trials. Prev Chronic Dis. 2017;14:E89.
- 178. Reuland DS, Malo TL, Brenner AT. Shared decision making rarely happens for lung cancer screening – reply. JAMA Intern Med. 2019;179(1):122–3.

Colorectal Cancer Screening

Veronica Sikka, Raaj K. Popli, and Edward P. Cutolo Jr.

Case Study

Patient is a 68-year-old male presenting to the emergency department (ED) with hematemesis with a 30 pound unintentional weight loss in the last 2 months. His hemoglobin is 10 with associated lightheadedness. His previous hemoglobin last month was 14. He is also complaining of low back pain. He reports never having a colonoscopy.

Upon presentation, the patient is given IV fluids and an NG tube is placed. Additional labwork is pending. A CT of the abdomen reveals a large obstructing mass in the cecum with dilated small bowel. Multiple liver metastases are seen affecting most lobes as well as lytic lesions in the lumbar spine. What is this patient's prognosis?

If the patient had received his routine colonoscopy, this cancer could have been detected at the polyp stage. Treatment options include (1) resection of the primary lesion; (2) palliative radiotherapy to the vertebral metastasis; and/or (3) systemic chemotherapy with biological therapy. His expected prognosis is not promising with a median survival of 24 months with systemic treatment.

V. Sikka (🖂)

Background

Colorectal cancer (CRC) is the fourth most commonly diagnosed cancer in the United States and the third most common cause of death for both men and women after lung and prostate for men and lung and breast for women. The American Cancer Society estimates that in 2020, 104,610 people will be diagnosed with CRC and 53,200 people will die from the disease alone [1]. Fortunately, the incidence of CRC has declined steadily in recent years and is largely attributed to the detection and removal of precancerous polyps with CRC screening [2, 3].

Although the overall incidence is declining, incidence in patients under the age of 50 is still on the rise as screening is not as common [4]. Surgery is considered the first-line therapy for CRC and is generally elective. Chemotherapy and radiation are also treatment options in more advanced stages. Unfortunately, patients with CRC may present to the emergency department (ED) with complications such as perforation, hemorrhage, and obstruction, as well as general complications secondary to chemotherapy and radiation [5]. This chapter focuses on (1) the diagnosis of CRC in the ED and (2) the ED recognition and management of patients with CRC-related complications.

The lifetime risk of obtaining colorectal cancer is 1 in 23 (4.4%) for men and 1 in 25 (4.1%) for women, with a higher predominance in men compared to women. CRC incidence is 30% higher in men. Lifetime risk is similar in men and women despite higher incidences in men. This is attributed to higher life expectancies in women. In comparing 5-year age groups, the incidence rate almost doubles until age 50, after which it increases by approximately 30%. Interestingly, patients diagnosed with CRC are younger today than ever before. In the early 2000s, the median age at diagnosis was 72. Today it is 66 years old. This is due to an increase in CRC screening over the years [3, 6].



Clinical Contact Center, VA Sunshine Healthcare Network (VISN 8), St. Petersburg, FL, USA

Associate Professor of Emergency Medicine, School of Medicine, University of Central Florida, Orlando, FL, USA e-mail: veronica.sikka@va.gov

R. K. Popli Department of Internal Medicine, University of Central Florida College of Medicine, Orlando, FL, USA

E. P. Cutolo Jr. Veterans Integrated Service Network 8 (VISN 8), St. Petersburg, FL, USA

Diagnosis of CRC

The diagnosis of CRC is not straightforward. First, the cancer can be discovered at any stage during progression, from asymptomatic cancer identified by screening through consultation with a general practitioner or gastroenterologist to presentation as a surgical emergency (usually with bowel obstruction or perforation) [7, 8]. If diagnosis is delayed and surgery is emergent or palliative in nature, it is associated with a substantially elevated risk of mortality [9], especially among the elderly [10]. However, if detected early, the 5-year survival rate is 90% while still localized (i.e., confined to the bowel wall), 68% for regional disease (i.e., lymph node involvement), and only 10% if distant metastases are present [1, 11, 12].

Risk for CRC is increased by genetic mutations (i.e., familial adenomatous polyposis, Lynch syndrome, juvenile polyposis, and Peutz-Jeghers syndrome), personal history (i.e., irritable bowel syndrome, Crohn's disease, and ulcerative colitis), and lifestyle factors (i.e., smoking, alcohol consumption, and diets high in fat and low in fiber). Earlystage CRC does not usually present with symptoms. Therefore, screening is necessary to detect cancer at earlier stages. Patients with advanced disease may present with changes in bowel habits, blood in the stool, weakness, fatigue, shortness of breath, signs of intestinal obstruction (i.e., bloating, fullness, cramps, and pain), unexplained weight loss, pain with defecation, and thin stools. In addition to CRC, the differential diagnosis for these symptoms includes hemorrhoids, infection, and inflammatory bowel disease.

The American Cancer Society [13] provides recommendations for guideline CRC treatment by TNM (tumor, nodes,

metastasized) stage. Specifically, resection is recommended for stage 0, I, II, and III CRC, and chemotherapy is guideline care for stages III and IV of CRC. Surgery is the most common treatment for CRC with the usual operation being either a segmental resection, partial colectomy, or diverting colostomy in the case of obstruction. Especially for CRCs that have not spread, surgical removal may be curative [14]. The choice of operation depends mainly on the site of the disease (left-sided versus right-sided), the patient's physical condition, nutritional status, and age. The treatment for right-sided lesions is a right hemicolectomy. However, treatment of leftsided lesions is still undecided. There are many therapeutic options such as primary or staged resections, Hartmann's procedure, subtotal colectomy, or colostomy. Other therapies involve nonoperative techniques such as laser therapy, colonic stenting, emergency endoscopy, and comfort measures.

Table 13.1 correlates the stages of CRC with the TNM categories and their associated management.

Trends in CRC incidence and mortality reveal overall declining rates, which have been attributed to reduced exposure to risk factors, early detection through screening, and prevention through polypectomy and improved treatment [15]. However, studies show a majority of US adults are not receiving age- and risk-appropriate screening or have never been screened [11, 16–19]. Among CRC patients, only 39% are actually diagnosed at an early stage, mostly due to these low screening rates [1]. Significant delays in screening translate into worse outcomes in terms of stage of cancer at diagnosis, ability for curative treatment, likelihood of recurrence, and survival, especially among the elderly [20].

Current recommendations for colorectal cancer screening from the US Preventive Services Task Force are presented in Table 13.2.

Table 13.1 Correlation between TNM categories and stage for CRC

Stage	TNM category	Interpretation	Colon cancer management	Rectal cancer management
0	Tis, N0, M0	Early-stage cancer where the cancer is limited to the mucosa of the colon or rectum (Tis). No lymph node involvement (N0) or distant spread (M0)	Surgery	Surgery
Ι	T1–T2, N0, M0	The cancer has grown through the mucosa into the submucosa (T1) or muscularis propria (T2). No lymph node involvement (N0) or distant spread (M0)	Surgery	Surgery \pm radiation
Π	T3–T4, N0, M0	T3–T4 stage with no spread to lymph nodes (N0) or distant sites (M0)	Surgery \pm chemotherapy	Surgery + chemotherapy + radiation
III	Any T, N1–N2, M0	Any <i>T</i> stage with spread to $1-3$ (N1) or four or more (N2) regional lymph nodes. No distant spread (M0)	Surgery + chemotherapy \pm radiation	Surgery + chemotherapy + radiation
IV	Any T, any N, M1	The cancer can be any T stage and any N stage and has spread to distant sites such as the liver, lung, peritoneum, or ovary (M1)	Surgery + chemotherapy ± other treatments (RFA, cryosurgery)	Surgery + chemotherapy + radiation

Adapted from the American Cancer Society (2020) [13]

TNM Tumor, nodes, metastasized, RFA radiofrequency ablation

Population	Recommendation	Grade
Adults, beginning at age 50 years and continuing until age 75 years	The USPSTF recommends screening for colorectal cancer using fecal occult blood testing, sigmoidoscopy, or colonoscopy in adults, beginning at age 50 years and continuing until age 75 years. The risks and benefits of these screening methods vary	The USPSTF recommends the service. There is high certainty that the net benefit is substantial
Adults age 76–85 years	The USPSTF recommends against routine screening for colorectal cancer in adults 76–85 years of age. There may be considerations that support colorectal cancer screening in an individual patient	The USPSTF recommends against routinely providing the service. There may be considerations that support providing the service in an individual patient. There is at least moderate certainty that the net benefit is small
Adults older than age 85 years	The USPSTF recommends against screening for colorectal cancer in adults older than age 85 years	The USPSTF recommends against the service. There is moderate or high certainty that the service has no net benefit or that the harms outweigh the benefits
Computed tomographic colonography and fecal DNA testing as screening modalities	The USPSTF concludes that the evidence is insufficient to assess the benefits and harms of computed tomographic colonography and fecal DNA testing as screening modalities for colorectal cancer	The USPSTF concludes that the current evidence is insufficient to assess the balance of benefits and harms of the service. Evidence is lacking, of poor quality, or conflicting, and the balance of benefits and harms cannot be determined

Table 13.2 Summary of US Preventive Services Task Force recommendations

Source: https://www.uspreventiveservicestaskforce.org/uspstf/recommendation/colorectal-cancer-screening, accessed 11 Jul 2020

The overall relative survival rate for CRC is 65% at 5 years following diagnosis and 58% at 10 years [6]. Forty percent of CRCs are diagnosed at a local stage, for which the 5-year survival rate is 90%. Thirty-six percent of CRCs are diagnosed at regional stage, for which the 5-year survival rate is 70%. Twenty percent of CRCs are diagnosed at a distant stage, for which the 5-year survival rate is 12%.

CRC Diagnosis in the ED

The number of patients presenting to the ED who are subsequently diagnosed with cancer in the ED has increased [21, 22]. In a comparison of patients diagnosed with cancer in the ED versus other settings, patients in the former category were found to be older, have Medicare or Medicaid, have stage IV cancer, and exhibit more symptoms [23]. A population-based study of 11,023 patients in Connecticut reported patients admitted from the ED with a CRC diagnosis were usually older (75+) and African American. An ED admission status was a significant predictor of distant stage in all patients [24]. A study of 151 patients in the United Kingdom examined the pathways to diagnosis of CRC. Despite considerable investment by the UK National Health Service in cancer diagnostic services for primary and specialty practices, 26% of patients had an emergency diagnosis [25].

Diggs et al. [26] focused on predictors and the associated burden of emergency CRC resection (E-CCR), which has Table 13.3 Emergent symptoms suggestive of colorectal cancer

Bleeding from the rectum
Blood in the stool or in the toilet after having a bowel movement
Dark or black stools
A change in the caliber or shape of the stool (i.e., narrow stools)
Cramping or discomfort in the lower abdomen
An urge to have a bowel movement when the bowel is empty
Constipation or diarrhea that lasts for more than a few days
Decreased appetite
Unintentional weight loss
Weakness and fatigue secondary to anemia

been defined as the "clearest evidence on an individual level for a failure of screening" [27]. This cross-sectional study of over 120,000 discharges nationally focused on patients who underwent the procedure emergently, finding older patients dually eligible for Medicare and Medicaid were at higher risk for E-CCR. There was also a threefold increase in hospital mortality, longer lengths of stay, and more than \$250 million in additional hospital charges associated with E-CCR. This study was limited in its focus on one type of cancer and a particular procedure associated with CRC.

Early CRC may present to the ED with vague to no symptoms, which further emphasizes the importance of screening. Symptoms that may suggest CRC and the need for additional screening if not already diagnosed with CRC are presented in Table 13.3.

A positive family history of colon cancer should also raise suspicion for CRC on the differential. Findings on physical exam include grossly positive or guaiac stools. It is important to get basic labs (i.e., CBC and BMP) since blood loss from the cancer leads to anemia, specifically iron deficiency anemia. In the ED, a CT of the abdomen with contrast can help locate and characterize a mass. Timely evaluation of symptoms consistent with CRC is essential, even for adults younger than age 50. If stable, the patient can be discharged with referral to a gastroenterologist for colonoscopy and/or surgeon if a mass is found.

Oncologic Emergencies Associated with CRC

More common is the management of the complications of patients already diagnosed with CRC, which include bowel obstructions, perforations, rectal bleeding, and complications secondary to chemotherapy and radiation. The sections below describe the relationship between CRC and the respective complications and associated history and clinical findings with the appropriate ED management [28].

Bowel Obstruction

As a tumor grows, it may bleed or cause obstruction of the colon. Intestinal obstruction can occur when tumor growth has invaded the lumen of the large intestine. Up to 20% of colon cancer in some series will present as bowel obstruction. This is more likely to occur in the left colon because it is narrower, with the splenic flexure particularly vulnerable [29].

Patients may present with diffuse, colicky abdominal pain, nausea, vomiting, and abdominal distension. They also may have decreased to no bowel movements and flatus. On advance stages, be hypotensive, tachycardic, and febrile. Labs such as a lactate, CBC, and BMP may be helpful and reveal a metabolic acidosis. Imaging includes an acute series that reveals multiple air-fluid levels, more than 3 cm of dilatation of the small bowel, and/or more than 3 mm thickening of the small bowel (Fig. 13.1).

An abdominal CT can also be very helpful in distinguishing between partial versus complete bowel obstructions as well as to assist in identifying the anatomic location of obstruction (Fig. 13.2). Intussusception primary or meta-



Fig. 13.2 Apple-core obstructing colon cancer following a barium enema demonstrated on an abdominal CT (axial). Note that the contrast material does not pass through the lesion retrograde and the upstream bowel appears to be distended



Fig. 13.1 (a) Small bowel obstruction on a KUB x-ray due to metastatic lobular breast cancer. (b) Coronal CT image revealing a segmental stricture in the right lower quadrant with thickened enhancement of the small bowel wall

static deposits to the bowel can contribute to obstruction (Figs. 13.3 and 13.4).

ED management includes symptomatic treatment with IV fluid boluses, antiemetics, and analgesia. An NG tube may be placed for a significant obstruction, especially if vomiting, and keeping the patient NPO for bowel rest. These patients require admission with gastroenterology and surgery consulting.



Fig. 13.3 Small bowel obstruction from intussusception secondary to melanoma metastasis, located in the right abdomen in front of the right kidney as demonstrated on an abdominal CT (axial)



Fig. 13.4 Intussusception from a primary cecal tumor as demonstrated by an abdominal CT (coronal)

Perforation

Patients with CRC may also present with bowel perforations as the CRC invades through the colon wall in more advanced stages. Patients may present with an acute onset of severe abdominal pain possibly associated with near or complete syncope. They are often unable to localize the pain but report worsening pain with any movement (parietal pain). Anorexia is common, but vomiting is often uncommon. On exam, the patient may have acute peritonitis with a rigid abdomen and rebound tenderness. Critical studies include an upright chest x-ray to ensure no air under the diaphragm. Free air can be seen in 70–94% of cases. A CT abdomen/pelvis would be the definitive study if the CXR is inconclusive.

ED management includes IV fluid resuscitation and antibiotics (i.e., 3.375 g of IV Zosyn). Immediate surgical consults are required with patients with perforation.

Rectal Bleeding

Patients with CRC can present with blood in their stools in the setting of recent changes in their bowel habits (i.e., constipation). In general, cancers of the ascending colon tend to be larger and more frequently bleed. Cancers of the descending colon tend to be smaller and more obstructive. Predominant constitutional symptoms include anorexia, fatigue, weight loss, and presyncope. Patients can present from asymptomatic rectal bleeding to ill-appearing with pale conjunctivae, hepatomegaly (secondary to liver metastasis), abdominal or rectal mass, and/or guaiac-positive stools.

Labs include a lactate, CBC, BMP, and coagulation studies. A CBC often reveals microcytic anemia. A BMP can be indicative of an anion gap (lactic) acidosis that is secondary to hypoperfusion. Coagulation studies should be ordered if the patient is anticoagulated or has liver disease.

The ED management depends on the severity of the rectal bleeding. With significant bleeding, two large bore IV lines should be established, and the patient should be volumeresuscitated with normal saline and cross-matched for 2–4 units of blood. If the patient is anticoagulated, FFP may be required to reduce the INR from 1.5 to 2.5. Vitamin K may be needed if bleeding continues despite FFP. General Surgery should be consulted if significant bleeding or obstructive symptoms exist. If stable with an occult lower GI bleed, outpatient oncologic work-up may be appropriate.

Complications Secondary to Chemotherapy and Radiation

Patients in stages 1 and above of CRC may require chemotherapy and radiation which can present to the ED as severe nausea and vomiting. Chemotherapy often causes symptoms 2–3 days after treatment. The emergency medicine approach is dependent on the patient's clinical status. If they appear significantly dehydrated, labs should be drawn to rule out any electrolyte abnormalities. The patient may require IV hydration and antiemetics. The final ED disposition is dependent on the patient's clinical status (i.e., orthostatic, able to tolerate PO, etc.), and the patient's gastroenterologist, hematologist, and/or surgeon should be consulted.

Conclusion

Despite the decreasing incidence of CRC, emergencies secondary to this deadly cancer still exist. It is important for the ED physician to be able to recognize the signs and symptoms that may hint at a new CRC diagnosis as well as how to manage complications in patients with pre-existing CRC. The approach is multidisciplinary with consultation of gastroenterology, hematology/oncologic, and surgery depending on the patient's presentation; however, most important is the emergent recognition and stabilization of these often complex patients.

Acknowledgments We are thankful for the images provided by Drs. Fergus Coakley and Elena Korngold, Department of Diagnostic Radiology, Oregon Health & Science University, Portland, Oregon.

We express additional thanks for Dr. Charles R. Thomas, Jr., for providing critical editorial assistance during the final stages of the chapter.

References

- American Cancer Society. Cancer facts & figures 2020. Atlanta: American Cancer Society; 2020.
- Edwards BK, Ward E, Kohler BA, Eheman C, Zauber AG, Anderson RN, et al. Annual report to the nation on the status of cancer, 1975–2006, featuring colorectal cancer trends and impact of interventions (risk factors, screening, and treatment) to reduce future rates. Cancer. 2010;116:544–73.
- American Cancer Society. Colorectal cancer: facts & figures 2020– 2022. Atlanta: American Cancer Society; 2020.
- Surveillance, Epidemiology, and End Results (SEER) program stat database: NAACR incidence–CiNA analytic file, 1995–2010, for expanded races, custom file with county, ACS facts and figures projection project, North American Association of Central Cancer Registries, 2013.
- Cuffy M, Abir F, Audisio RA, Longo WE. Colorectal cancer presenting as surgical emergencies. Surg Oncol. 2004;13(2–3):149–57.
- Howlader N, Noone AM, Krapcho M, Miller D, Brest A, Yu M, et al., editors. SEER cancer statistics review, 1975–2016. Bethesda: National Cancer Institute; 2018, https://seer.cancer.gov/ csr/1975_2016/, based on November 2018 SEER data submission, posted to the SEER web site, April 2019.
- Kemppainen M, Raiha I, Rajala T, Sourander L. Characteristics of colorectal cancer in elderly patients. Gerontology. 1993;39(4):222–7.
- Majumdar SR, Fletcher RH, Evans AT. How does colorectal cancer present? Symptoms, duration, and clues to location. Am J Gastroenterol. 1999;94(10):3039–45.

- McArdle CS, Hole DJ. Emergency presentation of colorectal cancer is associated with poor 5-year survival. Br J Surg. 2004;91(5):605–9.
- Koperna T, Kisser M, Schulz F. Emergency surgery for colon cancer in the aged. Arch Surg. 1997;132(9):1032–7.
- 11. Levin B, Lieberman DA, McFarland B, Smith RA, Brooks D, Andrews KS, et al. Screening and surveillance for the elderly detection of colorectal cancer and adenomatous polyps, 2008: a joint guideline from the American Cancer Society, the US Multi-Society Task Force on Colorectal Cancer, and the American College of Radiology. CA Cancer J Clin. 2008;58:130–60.
- Ries LA, Melbert D, Krapcho M, Mariotto A, Miller BA, Feuer EJ, et al. SEER cancer statistics review, 1975–2004. Bethesda: National Cancer Institute; 2007.
- American Cancer Society Guideline for Colorectal Cancer Screening. Last medical review: 29 Jun 2020. Last revised: 29 Jun 2020. https://www.cancer.org/cancer/colon-rectal-cancer/ detection-diagnosis-staging/acs-recommendations.html. Accessed 1 Jul 2020.
- Guillem JG, Paty PB, Cohen AM. Surgical treatment of colorectal cancer. CA Cancer J Clin. 1997;47(2):113–28.
- Espey DK, Xu XC, Swan J, Wiggins C, Jim MA, Ward E, et al. Annual report to the nation on the status of cancer, 1975–2004, featuring cancer in American Indians and Alaska Natives. Cancer. 2007;110:2119–52.
- Breen N, Wagener DK, Brown ML, Davis WW, Ballard-Barbash R. Progress in cancer screening over a decade: results of cancer screening from the 1987, 1992, and 1998 National Health Interview Surveys. J Natl Cancer Inst. 2001;93(22):1704–13.
- Jemal A, Siegel R, Ward E, Hao Y, Xu J, Murray T, et al. Cancer statistics 2008. Cancer. 2008;58:71–96.
- Meissner HI, Breen N, Klabunde CN, Vernon SW. Patterns of colorectal cancer screening uptake among men and women in the United States. Cancer Epidemiol Biomark Prev. 2006;15(2):389–94.
- Smith RA, Cokkinides V, Eyre HJ. Cancer screening in the United States, 2007: a review of current guidelines, practices, and prospects. CA Cancer J Clin. 2007;57(2):90–104.
- Gornick ME, Eggers PW, Riley GF. Associations of race, education, and patterns of preventive service use with stage of cancer at time of diagnosis. Health Serv Res. 2004;39(5):1403–27.
- 21. Sikka V, Ornato JP. Cancer diagnosis and outcomes in Michigan EDs vs other settings. Am J Emerg Med. 2012;30(2):283–92.
- Swenson KK, Rose MA, Ritz L, Murray CL, Adlis SA. Recognition and evaluation of oncology-related symptoms in the Emergency Department. Ann Emerg Med. 1995;26(1):12–7.
- Weithorn D, Arientyl V, Solsky I, Umadat G, Levine R, Rapkin B, et al. Diagnosis setting and colorectal cancer outcomes: the impact of cancer diagnosis in the emergency department. J Surg Res. 2020;255:164–71.
- Polednak AP. Inpatient hospital admission through an emergency department in relation to stage at diagnosis of colorectal cancer. Cancer Detect Prev. 2000;24(3):283–9.
- 25. Barrett J, Jiwa M, Rose P, Hamilton W. Pathways to the diagnosis of colorectal cancer: an observation study in three UK cities. Fam Pract. 2006;23(1):15–9.
- Diggs JC, Xu F, Diaz M, Cooper GS, Koroukian SM. Failure to screen: predictors and burden of emergency colorectal cancer resection. Am J Manag Care. 2007;13(3):157–64.
- Smothers L, Hynan L, Fleming J, Turnage R, Simmang C, Anthony T. Emergency surgery for colon carcinoma. Dis Colon Rectum. 2003;46(1):24–30.
- Marx J, Hockberger R, Walls R. Rosen's emergency medicine concepts and clinical practice. 8th ed. Philadelphia: Saunders; 2014.
- 29. Phillips RK, Hittinger R, Fry JS, Fielding LP. Malignant bowel large bowel obstruction. Br J Surg. 1985;72(4):296–302.



Melanoma Screening

Wei-Lynn Chong, Thomas M. Pitney, and Michael Sinnott

Case Study

A 23-year-old male is brought in by ambulance after having a syncopal episode on the beach. He is in Australia from the UK over the summer backpacking with friends. Collateral from the ambulance reports that he and his friends have spent the day on the beach and had consumed multiple beers. Temperatures outside have reached 38 degrees Celsius and there was no shade on the beach. He was playing volleyball and had a witnessed collapse where he lost consciousness. His friends placed him on his side and a bystander called the ambulance. There were no witnessed shaking movements or incontinence. After a couple of minutes, he regained consciousness and has Glasgow Coma Scale score of 15 since that time.

He reports a few days of heavy alcohol intake prior to today and has spent the weekend on the beach in the sun. He has no other medical conditions and denies any regular medications or recreational drug use. His examination reveals blistering sunburn to his face, trunk and upper limbs. His examination reveals normal vital signs except for a postural tachycardia, dry mucus membranes and an incidental finding of multiple melanocytic naevi with a large, irregular lesion on his lower back. Investigations of a bedside electrocardiogram, full blood count and biochemistry panel reveal a mild prerenal acute kidney injury consistent with dehydration. The impression is that the patient has had a syncopal

Emergency Department, Princess Alexandra Hospital, Brisbane, QLD, Australia e-mail: weilynn.chong@health.qld.gov.au

T. M. Pitney Department of Dermatology, The Mater Public Hospital, Brisbane, QLD, Australia

M. Sinnott The University of Queensland Medical School, Brisbane, QLD, Australia event secondary to dehydration from sunburn, heat exhaustion and excess alcohol intake. The plan for his management in the emergency department includes IV fluid therapy, application of soothing gels to his sunburn, teledermatology referral for review of the lesion on his lower back and education on sun protection and alcohol consumption. On discharge his vitals have normalised and he is feeling much better. He has an appointment with dermatology in 3 days for consideration of excision of the lesion on his back. He was given written and verbal advice about sun protection, including wearing sunscreen, a hat, sunglasses and clothing to protect himself from the sun when outside, as well as to avoid being in direct sunlight between 10 am and 2 pm.

Introduction

In both the USA and Australia, skin cancer is the most commonly diagnosed malignancy. Importantly it is also the most easily preventable and most successfully treated when diagnosed early [1]. Although less common than basal cell carcinoma (BCC) and squamous cell carcinoma (SCC), melanoma is more important to diagnose early due to its high metastatic potential and mortality risk. A melanoma is a malignant tumour arising from melanocytes. It is easily treatable if detected at an early stage. In situ lesions (confined to the epidermis) have a very low risk of metastatic spread. The main risk factor for metastasis and mortality is the depth of dermal invasion [2].

Given the importance of early detection, melanoma has been the focus of many primary prevention programmes in Australia, notably the "Slip, Slop, Slap" campaign of the 1980s, which encouraged the public to Slip on a shirt, Slop on sunscreen and Slap on a hat to limit excessive sun exposure [3]. There is evidence that this has been effective, with a noted age-specific decrease in incidence of melanoma in the 0–39 age group from a peak of 13 to 9.4 per 100,000. The same data reveal a decreased incidence of invasive melanomas in the under 55 group and a slower increase than prior in

W.-L. Chong (🖂)

K. H. Todd et al. (eds.), Oncologic Emergency Medicine, https://doi.org/10.1007/978-3-030-67123-5_14

the above 55 group [4]. Unfortunately, the proportion of thick melanomas at initial diagnosis has not changed over time. This is thought to reflect their highest incidence in older people, particularly men, to whom little specific advertising or primary prevention campaigns have been directed. This increasing incidence with age will become more relevant over time as populations age in the western world.

The burden of skin cancer in Australia is largely managed by public outpatient "general practitioner" physicians, with a smaller number being managed in public or private hospital outpatient clinics and very little being specifically managed by the urgent care or emergency physician. The importance of the emergency department (ED) in melanoma prevention and screening is in its valuable contact with people who may most benefit, such as those from lower socioeconomic groups. They may have poorer health literacy and have been shown to have poorer melanoma prognosis [5]. An unrelated ED visit for an emergency issue may be their sole point of contact with the medical profession for extended periods of time and an important opportunity to provide both primary prevention advice with sun protection, self-identification of suspicious lesion education and a secondary screening chance while addressing their primary presenting issue.

Opportunistic at-risk groups may include ED presentations for sunburn, or patients with melanoma risk factors such as a personal or family history of melanoma and other solar-related skin malignancies, solar damage, multiple naevi or immunosuppression. Other at-risk groups include young people without a primary healthcare physician. Identification of these groups with referral to specialist services when appropriate has demonstrated both improved outcomes for the patients and cost-effectiveness compared to standard community care. In a 2017 cost-effectiveness study, over a 10-year period, specialised surveillance services were estimated to result in a savings of \$6828 (in 2013 Australian dollars) and a gain of 0.31 quality-adjusted life-years per patient when compared to standard care [6].

Epidemiology

Melanoma has an incidence rate of 49 per 100,000 in the Australian population and a mortality rate of 6.2 per 100,000 (standardised for age). Melanoma accounts for 10% of all cancers diagnosed in Australia and for 3.8% of cancer deaths [1].

Risk factors for melanoma are both environmental and genetic. Environmental ultraviolet (UV) radiation exposure from the sun with repeated significant sunburns appears to carry the greatest risk [7]. Sunbed (tanning bed) use is considered an independent risk factor, particularly if first exposure occurs before the age of 35 [8].

Genetic factors may increase risk of melanoma. Phenotypic features associated with this risk are skin types having poor ability to tan and higher risk of sunburn, including those with fair complexion, blond or red hair and pale eyes. Large numbers of melanocytic naevi are also associated with increased risk of melanoma. Other clinical features indicating higher risk include skin changes associated with solar damage, such as freckling, solar elastosis, solar lentigines and keratoses [8–11]. Conversely, darker-skinned groups have a lower incidence of melanoma primarily as a result of photoprotection provided by increased epidermal melanin, which filters twice as much UV radiation as does that in the epidermis of fair-skinned groups [9]. As such melanoma demonstrates greater variation in incidence rates across different ethnic groups than that of most cancers [10].

Although skin malignancies occur less frequently in those with dark skin types, they suffer from higher rates of mortality and morbidity. This is likely due to delayed diagnosis [9]. Though UV radiation exposure is not as significant compared to those with high-risk phenotypic features, sun protection is still advised to protect from basal cell carcinoma development.

A personal or family history of melanoma in parents or siblings is associated with increased risk. Having a parent with multiple melanomas is a particularly strong risk factor [12].

Pathophysiology

UV radiation (wavelength 100–400 nm) can be classified as UVC (100–280 nm), UVB (280–315 nm) and UVA (315– 400 nm). All UVC and 90% of UVB are absorbed in the atmosphere. UV radiation has been implicated in the development of cutaneous melanoma. The precise mechanism of this pathogenesis is poorly understood; however, both UVB and UVA are implicated. UVB-induced mutations of the phosphatase and tensin homologue (PTEN) tumour suppressor gene, by direct DNA base pair damage, have been demonstrated in melanoma samples from patients with xeroderma pigmentosum [11]. UVA, which penetrates deeper into the dermis of the skin than UVB, can itself induce DNA damage through production of radical oxygen species and exerts an immunosuppressant effect which may also contribute to the development of melanoma [12].

Clinical Manifestations

Melanoma can present in a multitude of ways. Clinical features vary with melanoma subtype, tumour width and thickness, the anatomical site, the presence of regression or inflammation and even patient skin type (Table 14.1 [13, 14];

Table 14.1 Melanoma subtypes

Туре	Incidence	Clinical features
Superficial spreading (Figs. 14.1 and 14.2)	Most common subtype in fair skin, rare before 4th decade	50% develop in a pre-existing naevus, and there may be clinically contrast between these two components [13]. Generally macular in early stages but can develop nodular components as it progresses
Nodular (Fig. 14.3)	Incidence peaks in 5th to 6th decade of life	Most common on the trunk region. Has rapid vertical growth pattern
Lentigo maligna (Fig. 14.4).	Elderly population	Often progress very slowly through an in situ phase from a pre-existing solar lentigo, usually on a sun-damaged skin background. Increase in size, depth of colour and irregularity of pigment are clues that suggest malignant change
Acral	10% of melanoma	Appear as an irregular
lentiginous (Fig. 14.5)	on white skin but over 50% on darker skin types [14]	lentiginous pigmented area on the sole or palm of the hand
Subungual (Fig. 14.6)	Rare. Proportionally more frequent in darker skin types in 40–70 age group	Arise from the nail matrix and present as melanonychia (brown pigmentation of the nail). Extension of pigmentation onto the proximal nail fold (Hutchinson sign), heterogeneity of pigment colour, expansion of the width of pigment distribution and proximal growth are concerning features that separate from benign longitudinal melanonychia

Figs. 14.1, 14.2, 14.3, 14.4, 14.5 and 14.6). Lesions that are symptomatic, stand out from other naevi (the "ugly duck-ling" sign) or have a history of change should be examined closely. Dermoscopy is an essential tool to identify key features of malignant change but requires specific training in its use to be of benefit [15].

Commonly taught tools for both physicians and self-assessment of skin lesions (such as ABCDE mnemonic) are



Fig. 14.1 Dermoscopy image of Fig. 14.2a (superficial spreading melanoma)



Fig. 14.2 (a, b) Superficial spreading melanoma



Fig. 14.3 Amelanotic nodular melanoma



Fig. 14.5 Acral lentiginous



Fig. 14.4 Lentigo maligna



Fig. 14.6 Subungual melanoma

simple enough that patient education should be attempted in patients considered at risk [16]. This mnemonic reminds the patient to look for Asymmetry, Border Irregularity, Colour that is not uniform, Diameter (larger than a pea) and Evolving size, shape or colour.

Any of the above variants can also rarely be amelanotic, appearing as "skin coloured" or red, pink or purple discoloured areas with or without nodular components.

Teledermatology

Technological advancements in dermatology over the last two decades include the introduction of teledermatology into healthcare organisations worldwide to expand the availability of specialty services. Teledermatology is an innovative model of care and an alternative to traditional face-to-face specialist consultation. Primary referral physicians have greater and faster access to specialist dermatology opinion, and dermatologists are able to increase their work-flow efficiency through this method. There are two applications of teledermatology: (1) live interactive and (2) store and forward technologies; or a combination of both can be utilised [17]. Live interactive teledermatology involves live conferencing between the primary healthcare worker with the patient and specialist dermatologist. Store and forward technologies utilise still images and written clinical referral or over the phone referral to specialist dermatologists. Dermatology is a visual specialty and as such is highly suited to the use of digital images for diagnostic and disease management purposes [13].

The benefits of teledermatology are immense. Same-day access to dermatology opinion allows primary referral physicians, especially those in rural and regional areas, increased access to specialist services to improve their diagnosis, initiate treatment and reduce morbidity and mortality related to misdiagnoses or treatment delay [17, 18]. Successful programmes initiated in Queensland, Australia, revealed the highest number of external referrals came from a rural hospital, located 1200 km from the nearest dermatology clinic and that junior physicians represented the majority of referring clinicians [19]. Decreasing the number of second referrals to a dermatologist produces a reduction in cost burden with savings in patient travel times and wait times [18]. Teledermatology may also help address projected dermatology workforce shortages.

However, the limitations of teledermatology should be recognised. Live interactive teledermatology requires more initial financial investment compared to store and forward technology and is at the mercy of provider networks that may be unpredictable in rural areas and provide poor image quality [17]. Technological advances, increased consumer use of smart phones, increasing use of email, faster internet speeds and maturation of electronic health records may mitigate these downsides [20]. Medicolegal risks involving patient privacy and confidentiality are inherent in use of personal phones and emails as platforms to relay potentially sensitive patient information. Practitioners attempting to limit their legal liability in the event of a breach of privacy should demonstrate that reasonable measures were taken to protect patient information. This includes obtaining patient consent before using teledermatology, blocking auto-uploads of clinical images from clinician's devices, encrypting or passwordprotecting images before transmission, deleting any patient information from personal email accounts or devices after use and securely storing any video recording of consultations [13]. Use of hospital or hospital-approved camera devices, work emails and institution-approved messaging platforms is preferred over personal equipment.

Investigation and Diagnosis

Definitive diagnosis can be difficult to make clinically in some cases. Excisional biopsies of any suspicious lesions are necessary for definitive diagnosis. Often, biopsy is not logistically possible in a busy ED, and responsibility falls to the local general practitioner, dermatologist or surgeon. Clear communication between the ED and the local physician (ensuring precise documentation of the lesion so both physician and patient are aware of its location) and educating the patient on the importance of following up this appointment to have the lesion biopsied is essential. Disadvantaged groups may benefit from having these appointments organised for them to improve compliance. If admitted, seeking consultation from inpatient dermatology or surgical teams can expedite the investigation and management of suspected lesions.

If melanoma has been considered as a diagnosis, the patient should be examined for evidence of regional or systemic dissemination. This involves palpation of draining lymph nodes and exclusion of hepatosplenomegaly in the first instance. Urgent biopsy and appropriate investigation and treatment should occur.

Metastasis of melanoma occurs through regional lymph node disease in 63%, direct haematogenous spread in 24% and satellite deposits in 13% as their first site of metastasis [21].

Treatment

The recommended initial management of any suspected melanoma is complete excisional biopsy with 2 mm margin into the subcutis. Lesions that are small and appear minimally invasive may be amenable to techniques such as deep shave excision (saucerisation) or punch excisional biopsy; however, these may risk deep or lateral margin involvement and require skill and practice to perform reliably. Partial biopsy techniques should be avoided if possible.

The most important prognostic factor remains Breslow depth – the degree to which the lesion has penetrated into the skin. This is shown by diagram in Fig. 14.7.

The depth of subsequent wide local excision for melanoma is dependent on maximum Breslow thickness.



Fig. 14.7 Breslow thickness (Courtesy and with permission of the Melanoma Research Alliance, Washington, DC)

Table 14.2 Breslow thickness and 10-year survival [23]

Breslow thickness (mm)	Recommended clear surgical margin	10-year survival
In situ	5–10 mm	-
<1	1 cm	92%
1.01-2	1–2 cm	80%
2.01-4	1–2 cm (2 cm suggested when possible to achieve, particularly if concerning high-risk features)	63%
>4	2 cm	50%

Recommended clear margins for increasing depth are detailed in Table 14.2 [22, 23].

Metastatic risk and subsequent mortality are strongly linked to the Breslow depth. Those with in situ or very superficial melanoma (<0.78 mm) will have high (95–100%) 5-year survival rates. Thicker lesions or metastatic lesions predict poorer survival (see Table 14.2). The most recent classification guidelines also take into account other factors such as localised tumour characteristics (ulceration, mitotic rate), lymph node involvement, metastasis location and biochemical blood markers such as lactate dehydrogenase.

There is no uniform indication for performing a sentinel lymph node biopsy in patients diagnosed with cutaneous melanoma. Current Australian recommendations suggest that melanomas with >1 mm Breslow depth or >0.8 mm with high-risk features will benefit. High-risk features include presence of mitoses, ulceration or lymphovascular invasion or age <40 years old [24]. Sentinel node biopsy provides useful prognostic information, but is not recommended for all cases due to complications associated with the general anaesthesia and documented risks of seroma, infection, false negatives (higher in certain locations such as head and neck) and lymphedema.

The role of PET or CT scanning is dependent on stage of disease. Patients with asymptomatic Stages 1–2 (localised disease with no sentinel lymph node positivity if applicable for testing) are not recommended for further investigation, only routine follow-up with full skin checks every 6–12 months [25]. Patients with sentinel lymph node positivity or thick tumours with high-risk features should be considered for imaging on a case-by-case basis, or if specific symptoms suggestive of metastasis exist (in consultation with medical oncologic).

The Future of Diagnosis

Research involving artificial intelligence (AI) to develop algorithms that diagnose early melanoma as well (or better) than physicians has sparked much debate over the benefits and harm of incorporating this technology into clinical practice [26]. The potential benefits of AI are numerous; however, more research is needed, and governance needs to be established, prior to implementation of AI into clinical practice. Given the ethical and medicolegal issues involved, it will be important to maintain close clinical oversight when AI is employed. Its best current use may be as a supplement to clinician diagnosis. AI cannot substitute for the physician's ability to integrate historical data with physical examination findings in generating a differential diagnosis and management plan. Clinical assessment is a multifactorial, unpredictable and complicated process. However, with technological advancement this may change. The challenge of preventing diagnostic skill decay in AI-assisted dermatology must be recognised as well as the potential risk for layperson misuse of downloadable smartphone applications [27].

The diagnostic accuracy of AI is limited by the accuracy of histopathological diagnoses that feed into its algorithms and neural networks. While pathologist-interpreted diagnoses are considered the current gold standard, there is clear acknowledgement that interobserver variability exists, particularly with "borderline" lesions (Lallas, Elmore) [26, 28]. Further research is required before the integration of AI into mainstream clinical practice.

Primary Prevention

Numerous primary prevention strategies to reduce UV radiation exposure to the general population exist worldwide. From community awareness campaigns to workplace education, as well as legislation against tanning bed use, these policies have resulted in a higher early detection rates for melanoma.



Fig. 14.8 Can you spot a skin cancer? Community awareness campaign poster (Courtesy and with permission of SunSmart, Cancer Council Western Australia)

Sunscreen contains chemicals that absorb and reflect UV radiation. The sun protection factor (SPF) measures a sunscreen's effectiveness at protecting against sunburn [29]. Effective sun protection health campaigns target the population on an individual, workplace and community level. In Australia, one of the most successful health campaigns was launched by Cancer Council Australia, a nongovernment organisation, in 1981. Their "Slip, Slop, Slap" campaign has evolved into a multicomponent programme targeting both primary and secondary screening practices. It advocates regular skin checks at an individual level and provides advice and resources (e.g. factsheets and online courses) for schools, early childhood facilities, workplaces, sport groups, community events and health professionals, as well as local government. Figure 14.8 is of a poster targeting regular skin checks as part of the "Slip, Slop, Slap, Seek, Slide" sun protection campaign. In 2019, nearly 40 years after the launch of this health campaign, Australia achieved its highest rate of early melanoma detection, with over 90% of cases diagnosed as Stage 1 or 3 [1]. Melanoma has near 100% 5-year survival when diagnosed at Stage 1.

Sunbed (tanning bed) use is associated with a significant increase in melanoma risk. This risk increases with the number of tanning sessions and with initial exposure at a young age (<35 years) [30]. In 2003, the World Health Organisation published *Artificial Tanning Sunbeds: Risk and Guidance*, outlining the role of UV radiation from the sun and artificial sources in the development of skin cancer (https://www.who. int/uv/publications/sunbedpubl/en/). Subsequently, legislative changes in Brazil and Australia led to a ban on the commercial use of tanning beds.

Other sun protection regimes include municipal planning policies outlining the importance of access to services and facilities providing protection from solar UV radiation, such as shade provision in public places like bus stops, parks, playgrounds and sporting and recreational facilities. Signage at community events encouraging sun protection is also encouraged.

Health professionals prescribing medications known to cause photosensitivity such as certain antibiotics, antiarrhythmics and antihistamines should counsel patients to avoid direct sunlight while taking these medications as well as about the importance of SPF in sunblock choice. Similarly, care for patients presenting to the ED with UV radiation (i.e. sunburn) should include opportunistic counselling regarding sun protection in the future.

Secondary Screening

Secondary prevention or early detection by means of full skin examinations is recommended, especially for individuals at high risk of melanoma. A structured surveillance protocol with 6 to 12 monthly full skin examinations, supported by total body photography and dermoscopy, is proven to be beneficial both clinically and economically [6]. There is no general consensus on timing of first skin check and intervals between skin checks. Educating high-risk individuals to recognise and document suspicious lesions is recommended [31].

Full skin examinations in the ED are not routinely completed due to time constraints and lack of follow-up by the same physician. Identification of a suspicious lesion opportunistically in the ED warrants follow-up through referral to dermatology or to a general practitioner for consideration of biopsy. However, presentation to an ED by at-risk populations with poor health literacy may be the only opportunity to diagnose, educate and encourage urgent follow-up of high-risk individuals.

References

- Australian Institute of Health and Welfare. Cancer in Australia 2019. Cancer series no.119. Cat. no. CAN 123. Canberra: AIHW; 2019.
- Thompson JF, Scolyer RA, Kefford RF. Melanoma a management guide for GPs. Aust Fam Physician. 2012;41(7):470–3. PMID: 22762064.
- 3. Victoria Cancer Council. SunSmart victoria. Slip! Slop! Slide! Original SunSmart campaign 2013 [15 May 2020]. Available from: https://www.sunsmart.com.au/tools
- Curchin DJ, Harris VR, McCormack CJ, Smith SD. Changing trends in the incidence of invasive melanoma in Victoria, 1985– 2015. Med J Aust. 2018;208(6):265–9.
- MacKie RM, Hole DJ. Incidence and thickness of primary tumours and survival of patients with cutaneous malignant melanoma in relation to socioeconomic status. BMJ. 1996;312(7039):1125–8.
- Watts CG, Cust AE, Menzies SW, Mann GJ, Morton RL. Costeffectiveness of skin surveillance through a specialized clinic for patients at high risk of melanoma. J Clin Oncol. 2017;35(1):63–71.
- Gandini S, Sera F, Cattaruzza MS, Pasquini P, Picconi O, Boyle P, et al. Meta-analysis of risk factors for cutaneous melanoma: II. Sun exposure. Eur J Cancer. 2005;41(1):45–60.
- International Agency for Research on Cancer Working Group on artificial ultraviolet (UV) light and skin cancer. The association of use of sunbeds with cutaneous malignant melanoma and other skin cancers: a systematic review. Int J Cancer. 2007;120(5):1116–22. Erratum in: Int J Cancer. 2007;120(11):2526.
- Gloster HM Jr, Neal K. Skin cancer in skin of color. J Am Acad Dermatol. 2006;55(5):741–60; quiz 61–4.
- Ferlay J, Soerjomataram I, Dikshit R, Eser S, Mathers C, Rebelo M, et al. Cancer incidence and mortality worldwide: sources, methods and major patterns in GLOBOCAN 2012. Int J Cancer. 2015;136(5):E359–86.
- Wang Y, Digiovanna JJ, Stern JB, Hornyak TJ, Raffeld M, Khan SG, Oh KS, Hollander MC, Dennis PA, Kraemer KH. Evidence of ultraviolet type mutations in xeroderma pigmentosum melanomas. Proc Natl Acad Sci U S A. 2009;106(15):6279–84.
- Wang SQ, Setlow R, Berwick M, Polsky D, Marghoob AA, Kopf AW, et al. Ultraviolet a and melanoma: a review. J Am Acad Dermatol. 2001;44(5):837–46.
- Stevenson P, Finnane AR, Soyer HP. Teledermatology and clinical photography: safeguarding patient privacy and mitigating medicolegal risk. Med J Aust. 2016;204(5):198–200, e1.

- Kittler H, Pehamberger H, Wolff K, Binder M. Diagnostic accuracy of dermoscopy. Lancet Oncol. 2002;3(3):159–65.
- Desai A, Ugorji R, Khachemoune A. Acral melanoma foot lesions. Part 1: epidemiology, aetiology, and molecular pathology. Clin Exp Dermatol. 2017;42(8):845–8.
- Robinson JK, Perez M, Abou-el-Seoud D, Kim K, Brown Z, Liko-Hazizi E, et al. Targeted melanoma screening: risk self-assessment and skin self-examination education delivered during mammography of women. JNCI Cancer Spectrum. 2019;3(3):pkz047.
- Duong TA, Cordoliani F, Julliard C, Bourrat E, Regnier S, de Pontual L, et al. Emergency department diagnosis and management of skin diseases with real-time teledermatologic expertise. JAMA Dermatol. 2014;150(7):743–7.
- Nadkarni A, Cardwell LA, Mahler S, Feldman SR. Most common dermatology diagnoses to the emergency department in the paediatric population. J Eur Acad Dermatol Venereol. 2018;32(4):e144–6.
- Biscak TM, Eley R, Manoharan S, Sinnott M, Soyer HP. Audit of a state-wide store and forward teledermatology service in Australia. J Telemed Telecare. 2013;19(7):362–6.
- Lamel SA, Haldeman KM, Ely H, Kovarik CL, Pak H, Armstrong AW. Application of mobile teledermatology for skin cancer screening. J Am Acad Dermatol. 2012;67(4):576–81.
- Adler NR, Wolfe R, McArthur GA, Kelly JW, Haydon A, McLean CA, et al. Tumour mutation status and melanoma recurrence following a negative sentinel lymph node biopsy. Br J Cancer. 2018;118(10):1289–95.
- 22. Cancer Council Australia. Clinical guidelines network. What are the recommended safety margins for radical excision of primary melanoma? Last modified 30 Jan 2020. https://wiki.cancer.org.au/ australia/Guidelines:Melanoma. Accessed 13 May 2020.

- Balch CM, Gershenwald JE, Soong SJ, Thompson JF, Atkins MB, Byrd DR, et al. Final version of 2009 AJCC melanoma staging and classification. J Clin Oncol. 2009;27(36):6199–206.
- 24. The Australasian College of Dermatologists; Mar V, Kelly J, Soyer HP, Crotty K, Haydon A, Saw R, et al. Sentinel node biopsy in 2020: a guide for Australian dermatologists. 2020. https://www. dermcoll.edu.au/wp-content/uploads/2020/01/ACD-Guide-for-Sentinel-Node-Biopsy-in-2020-Jan-2020.pdf. Accessed 23 Aug 2020.
- Swetter SM, Tsao H, Bichakjian CK, Curiel-Lewandrowski C, Elder DE, Gershenwald JE, et al. Guidelines of care for the management of primary cutaneous melanoma. J Am Acad Dermatol. 2019;80(1):208–50.
- Lallas A, Argenziano G. Artificial intelligence and melanoma diagnosis: ignoring human nature may lead to false predictions. Dermatol Pract Concep. 2018;8(4):249–51.
- 27. Lim BCW, Flaherty G. Artificial intelligence in dermatology: are we there yet? Br J Dermatol. 2019;181(1):190–1.
- Elmore JG, Barnhill RL, Elder DE, Longton GM, Pepe MS, Reisch LM, et al. Pathologists' diagnosis of invasive melanoma and melanocytic proliferations: observer accuracy and reproducibility study. BMJ. 2017;357:j2813. Erratum in: BMJ. 2017 Aug 8;358:j3798.
- 29. Sarkany R. Sun protection strategies. Medicine. 2017;45(7):444-7.
- Boniol M, Autier P, Boyle P, Gandini S. Cutaneous melanoma attributable to sunbed use: systematic review and meta-analysis. BMJ. 2012;345:e4757.
- 31. Watts CG, Dieng M, Morton RL, Mann GJ, Menzies SW, Cust AE. Clinical practice guidelines for identification, screening and follow-up of individuals at high risk of primary cutaneous melanoma: a systematic review. Br J Dermatol. 2015;172(1):33–47.

Part III

Evaluation and Treatment



Acute Airway Obstruction

Christopher H. Chang and Lakshmi Mudambi

Case Study

A 65-year-old male presented with 2 months of large-volume hemoptysis. Prior medical history included Child-Pugh Class B cirrhosis with thrombocytopenia and hypothyroidism. He was an active smoker with a 110 packyear smoking history. Computed tomography showed a right upper lobe mass extending into the right mainstem and post-obstructive pneumonitis of the right upper lobe (Fig. 15.1, arrow). A positron emission tomography (PET) scan showed no distant or nodal metastasis (Fig. 15.2). Rigid bronchoscopy was performed to debulk and cauterize the tumor (before and after therapeutic bronchoscopy pictures), which controlled the hemoptysis. The patient was not a candidate for surgical resection due to his poor functional capacity and advanced hepatic disease, so he underwent definitive radiotherapy for a cT2aN0M0 squamous cell carcinoma of the right upper lobe.

Introduction

Malignant central airway obstruction (MCAO) is defined as a mechanical obstruction of the trachea or either mainstem bronchi impairing airflow to the lungs due to a malignant neoplasm.

C. H. Chang

Division of Pulmonary and Critical Care Medicine, Department of Medicine, Oregon Health and Science University, Portland, OR, USA

L. Mudambi (🖂)

Division of Pulmonary and Critical Care Medicine, Department of Medicine, Oregon Health and Science University, Portland, OR, USA

Division of Pulmonary and Critical Care, VA Portland Health Care System, Portland, OR, USA e-mail: Lakshmi.Mudambi@va.gov

Epidemiology

The most common cause of MCAO is a primary lung cancer with approximately 20–30% of patients with lung cancer developing symptoms related to central airway obstruction. MCAO can also occur due to metastasis from a solid organ malignancy, most commonly breast cancer, colorectal can-



Fig. 15.1 (a) Computed tomography revealed a right upper lobe mass extending into the right mainstem (*arrow*) and post-obstructive pneumonitis of the right upper lobe. (b) A positron emission tomography scan showed no distant or nodal metastasis

cer, and renal cancer [1]. It can result from external compression or direct invasion from tumors of adjacent structures such as the esophagus, thyroid, and mediastinum [2]. Other causes of MCAO include carcinoid tumors, which account for the majority of primary tumors distal to the main carina, and primary malignancies of the trachea [3], such as squamous cell carcinoma, adenoid cystic carcinoma, and mucoepidermoid carcinoma.

Classification

Bronchoscopy is used to characterize the appearance of the tumor and may be classified as an endoluminal (intrinsic), extraluminal (extrinsic), or mixed tumor. Endoluminal obstruction is caused by an exophytic lesion growing into the lumen of the airway. Extraluminal obstruction occurs when a tumor that is adjacent to an airway compresses the lumen. When both endoluminal and extraluminal obstruction occur from a tumor, it is classified as a mixed obstruction. The distinction between these types of obstructions is critical to developing a management strategy.

Presentation

The presentation of MCAO is variable, ranging from asymptomatic to catastrophic. Patients with MCAO present with a persistent dry or productive cough, dyspnea with exertion or at rest, wheezing, hemoptysis, or post-obstructive pneumonia. Stridor at rest suggests a tracheal diameter of less than 5 mm [4]. The diagnosis of MCAO can be elusive as many of these symptoms overlap with other diseases such as asthma, chronic obstructive airways disease, and pneumonia.

The development of dyspnea in an individual with MCAO is the result of an intricate interaction of multiple factors (Fig. 15.3). The sensation of dyspnea from a central airway obstruction is postulated to occur from the increased effort required to maintain airflow through a narrowed lumen [5]. Significant airway obstruction is defined as an endoluminal diameter of <50% of normal, but individuals who are otherwise healthy enough to generate sufficient negative intrathoracic pressure to overcome the increased airflow resistance are often able to tolerate much greater degrees of obstruction without symptoms. Factors such as the increased metabolic demand, cachexia from cancer, and the presence and severity of cardiac, pulmonary, and renal comorbidities also contribute to the sensation of dyspnea. Interestingly, some patients with significant MCAO and severe comorbidities may report being minimally symptomatic but only by limiting their activities.

Some symptoms may be dramatic and warrant urgent or emergent intervention. Hypoxemia that is refractory to supplementary oxygen or noninvasive ventilation can occur suddenly due to acute lobar atelectasis or after a period of slowly progressing dyspnea. Massive hemoptysis or hemoptysis associated with respiratory compromise creates urgent clinical situations that require immediate action. Post-obstructive pneumonias may not be responsive to antibiotics without relief of obstruction, can progress to sepsis, and preclude treatment with systemic chemotherapy.

Acute airway obstruction may also arise from complications related to prior treatment of MCAO. For example, radiotherapy is used extensively in the treatment of intrathoracic malignancies, and several reports of high-grade toxicities of the airways such as massive hemoptysis, airway necrosis, bronchial stenosis, and fistulas leading to central airway obstruction have been documented [6–9]. Airway stents can also lead to respiratory distress due to complications such as stent migration, stent fracture, and obstruction of the stent



Fig. 15.2 Before (*left*) and after (*right*) images from therapeutic bronchoscopy in which the mass tumor was debulked and cauterized



Fig.15.3 (a) Patient characteristics involved in the development of symptoms in malignant central airway obstruction (MCAO) (b) Tumor characteristics involved in the development of symptoms in MCAO

with mucous plugs or granulation tissue [10]. Although these situations cannot be purely designated as MCAO, it is reasonable to include them in this context since they present in the same patient population and can be life-threatening.

Evaluation

In addition to obtaining a complete history and performing a thorough clinical examination, radiographic imaging and bronchoscopy are useful tools to evaluate MCAO. Although plain radiographs are not very sensitive to detect MCAO, they may be the only imaging modality tolerated by a patient in respiratory distress. Plain radiographs can help determine laterality of the malignancy and show lobar or lung collapse, compression of the large airways, and lung masses. A computed tomography of the chest provides valuable information regarding the extent, location, morphology, and relationship of the tumor with surrounding structures, which is necessary for procedural planning [11, 12]. It is also useful to review prior imaging as the information obtained about the progression and rate of growth of a tumor can aid decision-making for both management and prognostication. Although flexible bronchoscopy is the gold standard diagnostic tool to assess MCAO, extreme caution must be employed as even gentle manipulation of a critical malignant airway obstruction can cause bleeding and edema resulting in an airway emergency, especially if immediate access to equipment and skilled personnel are unavailable to

manage these compilations. In patients with large anterior mediastinal masses, the use of sedatives and paralytic agents for intubation, bronchosocpy or other procedures can cause complete airway collapse and respiratory arrest due to extrinsic compression of the airways [13]. In such situations, it is important to establish an airway that traverses the collapsed segment and can be secured either by an awake bronchoscopic intubation, rapid sequence intubation or rigid intubation.

Management Overview

The approach to the management of MCAO begins with an assessment of symptoms and the extent of disease. Interventional pulmonology, thoracic surgery, otorhinolaryngology, and interventional radiology play collaborative and complementary roles in the management of MCAO depending upon location and presenting symptoms. In a true emergency, resuscitation and stabilization are the primary aims. In impending or complete obstruction without access to specialized care, awake bronchoscopic intubation with a smalldiameter, wire-reinforced endotracheal tube is an effective method to secure an airway in a critical malignant obstruction proximal to the main carina. Bronchoscopic guidance helps prevent excessive trauma to the tumor, thereby reducing the risk of hemorrhage while advancing the endotracheal tube past the obstruction. Rarely, in patients with dyspnea without significant hypoxemia, heliox can be used to reduce resistance to airflow across the malignant obstruction as a bridge to therapeutic bronchoscopy.

It is paramount to objectively define goals of an intervention in MCAO. Aside from addressing urgent or emergent clinical situations, therapeutic bronchoscopy has been shown to improve dyspnea, spirometry, walk distances, survival, and, most importantly, quality of life [14–17]. Additionally, patients with greater dyspnea and poorer functional status derive the most benefit from therapeutic bronchoscopy [18]. A large, prospective review from the multicenter AQuIRE registry showed that complication rates of therapeutic bronchoscopy are low, averaging 3.9%; however, there is significant variability between institutions (0.9–11.7%) [19]. Although overall survival in MCAO is dependent upon the underlying malignancy, therapeutic bronchoscopy can bring significant relief and comfort as a palliative intervention [20].

In operable patients with MCAO and minimal symptoms, it may be reasonable to proceed to surgery as the first step with the goal of curative treatment (Fig. 15.4). Sometimes, symptoms like large-volume hemoptysis necessitate therapeutic bronchoscopy to prevent complications such as respiratory failure before considering treatment for the underlying malignancy (Fig. 15.5). The lobar airways distal to the central obstruction should be patent and spared from tumor infiltration when contemplating therapeutic interventions to alleviate dyspnea and recanalize the lumen. This can usually be identified on cross-sectional imaging and bronchoscopy, using thin scopes to explore past the central lesion (Fig. 15.6). Patency of distal airways, either on CT or bronchoscopy, has been identified as predictors of a successful therapeutic bronchoscopy to relieve MCAO [18, 21]. Control of hemoptysis



Fig. 15.5 61-year-old male with large-volume hemoptysis and respiratory distress. Computed tomography of the chest showed a wellcircumscribed 2.5 cm nodule in the right upper lobe bronchus extending into the right mainstem (*arrow*). A rigid bronchoscopy was performed to debulk and cauterize the tumor. Two months later, he underwent a video-assisted thoracoscopic lobectomy for a pT1bN0M0 squamous cell carcinoma of the right upper lobe



Fig. 15.4 61-year-old male with a new left mainstem sessile polyp on computed tomography of the chest (**a**, *arrow*). Prior history of pT1N1M0 squamous cell carcinoma of the oropharynx treated with sur-

gery. Bronchoscopy revealed an endoluminal nodule and biopsy showed a non-small cell carcinoma with mucoepidermoid features (**b**). He underwent a thoracotomy with left mainstem sleeve resection

and an attempt to salvage contralateral lung are indications to intervene with therapeutic bronchoscopy, even when there is involvement of lobar bronchi (Fig. 15.7).

Management of massive or symptomatic hemoptysis is a harrowing clinical situation that can complicate MCAO. In symptomatic, alert patients maintaining their airway by coughing out blood, it is important to be completely prepared to assume control of the airway for intubation before paralyzing the patient as blood can quickly fill the trachea and mainstem bronchi leading to respiratory arrest. Visualization of the vocal cords can be difficult with direct laryngoscopy due to pooling of blood in the hypopharynx; video laryngoscopic and bronchoscopic visualization during intubation are useful adjuncts to secure an airway [22]. Tilting the patient toward the side of the bleed minimizes the blood flowing into the unaffected lung until the affected side can be isolated with an endobronchial blocker or contralateral mainstem intubation with a large endotracheal tube. If the cause of hemoptysis is known to be an endoluminal or mixed tumor and rigid bronchoscopy is readily available, the plan should be to proceed with rigid bronchoscopy as soon as possible. Bronchial artery embolization (BAE), a technique of selectively embolizing dilated or ecstatic bronchial arteries supplying tumors, must be considered if rigid bronchoscopy is not available, in cases of bleeding due to extraluminal



Fig. 15.6 73-year-old male with dyspnea and unresolving pneumonia of the right lower lobe. Computed tomography of the chest showed a mass in the right bronchus intermedius (**a**, *arrows*). Rigid bronchoscopy showed complete obstruction of the right bronchus intermedius with an endoluminal tumor (**b**). The right middle lobe and right lower lobe anterior segment were patent when the thin scope passed by the

central tumor. The tumor was debulked with improvement in dyspnea and right lower lobe pneumonia, and he received definitive chemotherapy and radiotherapy to cT3N0M0 of the right bronchus intermedius. No evidence of recurrence 2 years after treatment on computed tomography of the chest (**c**; *arrows show patent right bronchus intermedius and radiation changes*) MCAO or as a bridge to rigid bronchoscopy or surgery. BAE boasts high rates of immediate control of bleeding (>90%) [23, 24]. Recurrence after BAE occurs between 1% and 50% of cases and should be combined with defini-



Fig. 15.7 Elderly male with an endoluminal squamous cell carcinoma, cT2aN2M0 of the right upper lobe extending slightly into the right mainstem presenting with large-volume hemoptysis. After rigid bronchoscopy, debulking, and cautery of the tumor, the patient declined all treatment options. He presented one year later with dyspnea due to tumor infiltration of right mainstem, right bronchus intermedius, and main carina with extension across the midline over the left mainstem. Post-obstructive pneumonitis seen in the right upper lobe distal to the large right hilar mass (shown in computed tomography of the chest). He underwent rigid bronchoscopy with successful recanalization of the right mainstem and right bronchus intermedius followed by palliation radiation to prevent obstruction of the left mainstem. He passed away one year later

tive or palliative treatment of the underlying malignancy (surgery or radiation therapy) to prevent further episodes of hemoptysis [25]. Emergent surgical management of massive hemoptysis is associated with high mortality ranging from 15% to 25%. A single-center retrospective study of patients with massive hemoptysis showed that adopting BAE to control hemoptysis prior to surgery reduced mortality significantly [26]. Finally, the importance of correcting thrombocytopenia and coagulopathies as well as the reversal of antiplatelet agents and anticoagulants to control hemoptysis cannot be stressed enough.

Therapeutic Bronchoscopy: Tools

It is necessary to possess extensive knowledge about the unique characteristics of the tools used in therapeutic bronchoscopy to treat MCAO. Classic tools, such as the rigid bronchoscope, serve as foundational equipment through which interventions are performed safely. Other technologies have been adapted and customized for use in the airways from allied medical fields. The tools available can be classified into three broad categories: (i) mechanical tools, (ii) thermal tools, (iii) cold tools, and (iv) stents (Fig. 15.8).

Most MCAO require a variety of tools to reach the desired outcome of airway recanalization. To mitigate the perils of hemorrhage, endoluminal tumors are routinely cauterized prior to mechanical debulking or are debulked using tools that employ electrocautery. For extraluminal tumors, dilation followed by stenting is the only endoscopic method available to establish patency of the lumen.





Mechanical Debulking

Mechanical debulking can be accomplished using a variety of rigid or flexible tools. The main advantage of this modality is that it allows rapid debulking without requiring a reduction in the fraction of inspired oxygen. This section will focus on a few prominent mechanical tools to address MCAO.

Rigid Bronchoscopy

Invented in the late 1800s to remove foreign bodies from airways, the rigid bronchoscope is now considered a fundamental tool for therapeutic airway interventions. The rigid bronchoscope has dual purposes. It is a straight, hollow, metal tube with a large lumen allowing the use of single or multiple, flexible, and rigid equipment. Additionally, the distal end of the bronchoscope has ventilatory side ports to allow ventilation while performing interventions. It is recommended to use the rigid bronchoscope for most therapeutic bronchoscopic interventions. The rigid bronchoscope is inserted into the airways under direct visualization with a rigid telescope. Some common methods of ventilation with the rigid bronchoscope include jet ventilation (manual or automatic, high or low frequency), spontaneous-assisted ventilation, and controlled mechanical ventilation [27]. The tip of the rigid bronchoscope is beveled and can also be used to "core" out endoluminal tumors. A variety of rigid instruments such as the rigid forceps, rigid electrocautery, rigid dilator, and highvolume suction catheter can be inserted through the rigid bronchoscope. The rigid scope also acts as the conduit for the microdebrider and insertion or removal of airway stents, which will be discussed later in this chapter.

Microdebrider

The microdebrider consists of a rotating cutting blade within a rigid metal suction catheter that is connected to a power console. The suction pulls tumor tissue into the rotating blade for resection and removes blood clots and debris out of the plane of view. The bronchial microdebrider blade can be used to resect tumors from the trachea and proximal mainstem bronchi with a high degree of success [28–33]. In skilled hands, complications are rare, but this device should be used cautiously as perforation of the airway walls and surrounding vascular structures could occur.

Dilators

Airway dilators can be rigid (used through a rigid bronchoscope) or flexible (disposable). Dilators are used to create a lumen in extrinsic compression (extraluminal or mixed airway obstruction) prior to deployment of a stent in MCAO. Radial expanding balloon dilators are available in many sizes and can be used with or without wire guidance through the working channel of a therapeutic flexible bronchoscope.

Thermal Tools

Thermal energy delivered through thermal tools, such as electrocautery, laser, and argon plasma coagulation, produce a variety of effects upon tissue based on the temperature generated by the instrument. Coagulation is useful both prior to debulking to devascularize tumors and after to control hemostasis at the base of the tumor. Temperatures must reach 60 $^{\circ}$ C at the tissue to produce a coagulation effect. For all thermal tools, there is a risk of an airway fire, so the fraction of inspired air should be less than 0.4 during application.

Laser

Lasers are used for their abilities to cut, coagulate, and vaporize tissue with each type of laser varying in its efficacy of these properties. The depth of penetration is determined by the laser beam's wavelength, distance from the tissue, and tissue absorbance. The neodymium-doped yttrium aluminum garnet (Nd:YAG) laser is most commonly used in MCAO management as it has excellent coagulation and ablative properties. Given the deeper penetration of lasers compared to other tools, there is an increased risk of airway perforation and fistula formation. A large retrospective case series reported success rates as high as 90% when recanalizing an obstructed airway using laser [34].

Electrocautery

Electrocautery utilizes heat that is generated when an electric current is met with resistance. A "coagulation current" is created when electricity is applied in alternating currents. This produces bursts of high-voltage peaks of energy resulting in thermal damage and coagulation. When a "cutting current" or
sinusoidal non-modulated waveform is used, it creates a higher average power that allows for smooth cutting and minimal thermal damage. A "blended current" alternates between cutting and coagulating currents. Given the heterogeneity of tissue composition - and subsequently its resistance - the effects of electrocautery can be unpredictable. Complications are rare but may include airway perforation, airway fire, airway stenosis, hemorrhage from damage to vascular structures and deprogramming of an implanted cardiac device. Newer high-frequency generators allow for more precise control of the power delivered and an automatic safety switch if temperatures reach 100 °C to prevent the formation of exploding steam pockets. A variety of electrocautery tools exist (e.g., rigid and flexible probes, snares, forceps). The tip of electrocautery tools must be cleaned frequently to maintain sufficient direct contact with tissue. Retrospective studies have shown good efficacy of electrocautery in the management of malignant and benign CAO, with the largest study reporting significant recanalization by bronchoscopic exam in 94% of cases and improved aeration on computed tomography in 63% of cases [35].

Argon Plasma Coagulation

Argon plasma coagulation (APC) is a type of noncontact monopolar electrosurgery used in the airways for over two decades [36, 37]. As argon gas passes by a high-frequency electrode, it is ionized into a plasma field between the electrode and nearby tissue leading to coagulation and desiccation. Different probe tips may direct the argon plasma flow forward, laterally or circumferentially. The effect on tissue can be modified by changing the gas flow rate, power, and mode of energy delivery. APC acts more superficially than other thermal therapies, and the resulting coagulated and desiccated tissue adds another layer of resistance in reaching the deeper, unaffected tissue. Although rare, APC can cause fatal gas embolisms particularly at higher flow rates and longer pulse durations [38].

Cold Tools

Cryotherapy: Cryoprobe and Cryospray

Certain gases (such as carbon dioxide and nitrous oxide) rapidly expand when moving from a high-pressure to lowpressure environment causing a drop in temperature. This is called the Joule-Thompson effect and is the basis of cryotherapy. Of the two modalities of cryotherapy available for bronchoscopy, the cryoprobe is the preferred method used to quickly reestablish patency of the lumen in acute symptomatic endoluminal or mixed MCAO. In cryo-recanalization, the tip of the cryoprobe is placed in contact with the tumor and activated to freeze the tissue until it is adherent to the probe. The tissue, probe, and bronchoscope are gently pulled away en bloc and removed from the airways without touching the airway walls. The advantage of this method is that it does not necessitate the reduction of fraction of inspired air in patients who are hypoxemic. Several prospective and observational studies have demonstrated a 77–94% success rate of relieving the obstruction by 50–100%. The most common complication seen in these studies was mild to moderate bleeding controlled with APC and/or electrocautery (4–10% of cases) [1, 39].

Cryospray or cryoprobe can be used to treat inoperable lung cancer of the airways or in situ lesions of the mucosa by applying repeat freeze-thaw cycles of extremely low temperatures (below -40 °C) to tissue, which results in direct cellular damage by the repeated intra- and extracellular crystallization of the water as well as dehydration of the tissue. The cold temperatures also cause tissue ischemia through vasoconstriction, formation of microthrombi, and increased blood viscosity. This method results in recanalization slowly and cannot be used to quickly relieve symptoms of acute MCAO.

Airway Stents

Airway stenting is a palliative procedure used to maintain patency of an obstructed airway. Airway stents are broadly classified as silastic/silicone stents or metallic stents, based on the composed material. Metallic stents (self-expanding metalic stent (SEMS) or balloon expanding) may be bare (no indication in MCAO) or covered (partially or completely with silicone or polyurethane to prevent tumor growth into the stent. Silicone stents are placed and removed through a rigid bronchoscope. Metallic stents can be placed through an endotracheal tube, but when doing so, the rigid bronchoscope should be easily available for removal if the stent is not deployed correctly. The optimal stent for MCAO fits perfectly along the airway wall and is neither too large (which can incite granulation tissue due to mucosal ischemia) nor too small (which increases the risk for migration). It completely covers the tumor and extends 0.5 cm proximal and distal to the obstruction [40].

There are only a few indications for airway stents in MCAO. Firstly, symptomatic extraluminal or mixed obstruction can be stented as a bridge to definitive or palliative therapy (Fig. 15.9). All forms of recurrent, symptomatic MCAO (endoluminal, extraluminal, or mixed) that have exhausted treatment options such as radiation or surgery can be considered for airway stenting. The benefit of airway stenting in asymptomatic patients with MCAO is not clear and should be assessed on an individual basis. Finally, placing an airway stent in tracheoesophageal fistulas with airway obstruction or excessive secretions can improve symptoms but should be performed only after considering esophageal stenting first.

Several large retrospective studies on the use of stents in malignant tracheobronchial central airway obstruction have



Fig. 15.9 70-year-old male with mild hemoptysis and dyspnea with exertion. Computed tomography and a positron emission tomography (PET) scan showed a large left infrahilar mass causing complete obstruction of left mainstem bronchus and collapse of the left upper and lower lobes (**a**, *arrows*), showing a large hiatal hernia that was chronic

(b). Rigid bronchoscopy showed a mixed obstruction. The tumor was debulked, and a fully covered, self-expanding metallic stent was placed with resolution of dyspnea (c, *arrow*). Patient was treated with definitive chemotherapy and radiotherapy for a cT4N2M0 squamous cell carcinoma

reported on efficacy and safety of both silicone and selfexpanding metallic stents but also focus on complications which include migration, obstruction from secretion, granulation, infections, and ulcerations [10, 41]. An absolute contraindication for airway stent placement is extrinsic compression from a vessel since the risk of erosion and development of a catastrophic fistula is extremely high. Overall, an airway stent is considered a useful adjunct for MCAO that must be used judiciously [42].

Conclusion

Malignant central airway obstruction can seriously impair quality of life and be life-threatening. Even in the most advanced malignancies, therapeutic bronchoscopy offers palliation of symptoms caused by MCAO. It is paramount to consider goals of intervention carefully, weighing risks and benefits with patient preferences for management. A multimodality and multidisciplinary approach to MCAO can both address the acute manifestations and provide longterm control.

References

- Ernst A, Feller-Kopman D, Becker HD, Mehta AC. Central airway obstruction. Am J Respir Crit Care Med. 2004;169(12):1278–97.
- Marchioni A, Lasagni A, Busca A, Cavazza A, Agostini L, Migaldi M, et al. Endobronchial metastasis: an epidemiologic and clinicopathologic study of 174 consecutive cases. Lung Cancer. 2014;84(3):222–8.
- Wood DE. Management of malignant tracheobronchial obstruction. Surg Clin North Am. 2002;82(3):621–42.

- 4. Geffin B, Grillo HC, Cooper JD, Pontoppidan H. Stenosis following tracheostomy for respiratory care. JAMA. 1971;216(12):1984–8.
- Barach AL, Eckman M. The effects of inhalation of helium mixed with oxygen on the mechanics of respiration. J Clin Invest. 1936;15(1):47–61.
- Bezjak A, Paulus R, Gaspar LE, Timmerman RD, Straube WL, Ryan WF, et al. Safety and efficacy of a five-fraction stereotactic body radiotherapy schedule for centrally located non-small-cell lung cancer: NRG Oncology/RTOG 0813 Trial. J Clin Oncol. 2019;37(15):1316–25.
- Corradetti MN, Haas AR, Rengan R. Central-airway necrosis after stereotactic body-radiation therapy. N Engl J Med. 2012;366(24):2327–9.
- Ledoux B, Dupont M, Duplaquet F, Pirard L, Ocak S, Wanet M, Remouchamps V. Illustration of a fatal radiation-induced lung aneurysm: is central lung stereotactic radiotherapy to be banned? Cancer Radiother. 2019;23(8):926–9.
- Reveiz L, Rueda JR, Cardona AF. Palliative endobronchial brachytherapy for non-small cell lung cancer. Cochrane Database Syst Rev. 2012;12:CD004284.
- Ost DE, Shah AM, Lei X, Godoy MCB, Jimenez CA, Eapen GA, et al. Respiratory infections increase the risk of granulation tissue formation following airway stenting in patients with malignant airway obstruction. Chest. 2012;141(6):1473–81.
- Boiselle PM, Lee KS, Ernst A. Multidetector CT of the central airways. J Thorac Imaging. 2005;20(3):186–95.
- Kwong JS, Adler BD, Padley SP, Muller NL. Diagnosis of diseases of the trachea and main bronchi: chest radiography vs CT. AJR Am J Roentgenol. 1993;161(3):519–22.
- Erdos G, Tzanova I. Perioperative anaesthetic management of mediastinal mass in adults. Eur J Anaesthesiol. 2009;26(8):627–32.
- Ong P, Grosu HB, Debiane L, Casal RF, Eapen GA, Jimenez CA, et al. Long-term quality-adjusted survival following therapeutic bronchoscopy for malignant central airway obstruction. Thorax. 2019;74(2):141–56.
- Mahmood K, Wahidi MM, Thomas S, Argento AC, Ninan NA, Smathers EC, Shofer SL. Therapeutic bronchoscopy improves spirometry, quality of life, and survival in central airway obstruction. Respiration. 2015;89(5):404–13.
- Mohan A, Shrestha P, Madan K, Hadda V, Pandey RM, Upadhyay A, et al. A prospective outcome assessment after bronchoscopic interventions for malignant central airway obstruction. J Bronchology Interv Pulmonol. 2020;27(2):95–105.
- Oviatt PL, Stather DR, Michaud G, Maceachern P, Tremblay A. Exercise capacity, lung function, and quality of life after interventional bronchoscopy. J Thorac Oncol. 2011;6(1):38–42.
- 18. Ost DE, Ernst A, Grosu HB, Lei X, Diaz-Mendoza J, Slade M, AQuIRE Bronchoscopy Registry, et al. Therapeutic bronchoscopy for malignant central airway obstruction: success rates and impact on dyspnea and quality of life. Chest. 2015;147(5):1282–98.
- 19. Ost DE, Ernst A, Grosu HB, Lei X, Diaz-Mendoza J, Slade M, AQuIRE Bronchoscopy Registry, et al. Complications following therapeutic bronchoscopy for malignant central airway obstruction: results of the AQuIRE Registry. Chest. 2015;148(2):450–71.
- Catarata MJP, Saleiro S, Araújo VS. Outcomes of airway stents in the palliative care of patients with cancer. Am J Hosp Palliat Care. 2020;38(1):19–24. Epub ahead of print. Erratum in: Am J Hosp Palliat Care. 2020.
- Giovacchini CX, Kessler ER, Merrick CM, Gao J, Wang X, Wahidi MM, et al. Clinical and radiographic predictors of successful therapeutic bronchoscopy for the relief of malignant central airway obstruction. BMC Pulm Med. 2019;19(1):219.
- Hakanson E, Konstantinov IE, Fransson SG, Svedjeholm R. Management of life-threatening haemoptysis. Br J Anaesth. 2002;88(2):291–5.

- 23. Chen J, Chen LA, Liang ZX, Li CS, Tian Q, Yang Z, et al. Immediate and long-term results of bronchial artery embolization for hemoptysis due to benign versus malignant pulmonary diseases. Am J Med Sci. 2014;348(3):204–9.
- Wu XJ, Xing ZH, Tian J, Fan Y, Li YP, Sun JZ, et al. Long-term outcome of bronchial artery embolization in the treatment of massive hemoptysis. Zhonghua Jie He He Hu Xi Za Zhi. 2010;33(10):754–7. Article in Chinese.
- 25. Yoon W, Kim JK, Kim YH, Chung TW, Kang HK. Bronchial and nonbronchial systemic artery embolization for lifethreatening hemoptysis: a comprehensive review. Radiographics. 2002;22(6):1395–409.
- Shigemura N, Wan IY, Yu SC, Wong RH, Hsin MK, Thung HK, et al. Multidisciplinary management of life-threatening massive hemoptysis: a 10-year experience. Ann Thorac Surg. 2009;87(3):849–53.
- de Lima A, Kheir F, Majid A, Pawlowski J. Anesthesia for interventional pulmonology procedures: a review of advanced diagnostic and therapeutic bronchoscopy. Can J Anaesth. 2018;65(7):822–36.
- 28. Shepherd RW, Radchenko C. Bronchoscopic ablation techniques in the management of lung cancer. Ann Transl Med. 2019;7(15):362.
- Lunn W, Garland R, Ashiku S, Thurer RL, Feller-Kopman D, Ernst A. Microdebrider bronchoscopy: a new tool for the interventional bronchoscopist. Ann Thorac Surg. 2005;80(4):1485–8.
- Casal RF, Iribarren J, Eapen G, Ost D, Morice R, Lan C, et al. Safety and effectiveness of microdebrider bronchoscopy for the management of central airway obstruction. Respirology. 2013;18(6):1011–5.
- Tjahjono R, Chin RY, Flynn P. Tracheobronchial stents in palliative care: a case series and literature review. BMJ Support Palliat Care. 2018;8(3):335–9.
- Melendez J, Cornwell L, Green L, Casal RF. Treatment of large subglottic tracheal schwannoma with microdebrider bronchoscopy. J Thorac Cardiovasc Surg. 2012;144(2):510–2.
- Wahla AS, Khan I, Bellinger C, Haponik E, Conforti JF. Use of microdebrider bronchoscopy for the treatment of endobronchial leiomyoma. Clin Respir J. 2011;5(2):e4–7.
- Cavaliere S, Venuta F, Foccoli P, Toninelli C, La Face B. Endoscopic treatment of malignant airway obstructions in 2,008 patients. Chest. 1996;110(6):1536–42.
- Wahidi MM, Unroe MA, Adlakha N, Beyea M, Shofer SL. The use of electrocautery as the primary ablation modality for malignant and benign airway obstruction. J Thorac Oncol. 2011;6(9):1516–20.
- Morice RC, Ece T, Ece F, Keus L. Endobronchial argon plasma coagulation for treatment of hemoptysis and neoplastic airway obstruction. Chest. 2001;119(3):781–7.
- Sheski FD, Mathur PN. Endobronchial electrosurgery: argon plasma coagulation and electrocautery. Semin Respir Crit Care Med. 2004;25(4):367–74.
- Reddy C, Majid A, Michaud G, Feller-Kopman D, Eberhardt R, Herth F, Ernst A. Gas embolism following bronchoscopic argon plasma coagulation: a case series. Chest. 2008;134(5):1066–9.
- Hetzel M, Hetzel J, Schumann C, Marx N, Babiak A. Cryorecanalization: a new approach for the immediate management of acute airway obstruction. J Thorac Cardiovasc Surg. 2004;127(5):1427–31.
- 40. Mudambi L, Miller R, Eapen GA. Malignant central airway obstruction. J Thorac Dis. 2017;9(Suppl 10):S1087–110.
- 41. Lemaire A, Burfeind WR, Toloza E, Balderson S, Petersen RP, Harpole DH Jr, D'Amico TA. Outcomes of tracheobronchial stents in patients with malignant airway disease. Ann Thorac Surg. 2005;80(2):434–7; discussion 437–8.
- 42. Dutau H, Dumon JF. Airway stenting revisited: 30 years, the age of reason? J Bronchology Interv Pulmonol. 2017;24(4):257–9.

Central Nervous System

Prashant Rai, Akhil Shivaprasad, Ivo W. Tremont-Lukats, and Sudhakar Tummala

Introduction

In this chapter, we describe central nervous system (CNS) complications of cancer commonly presenting to the emergency department (ED), including altered mental status, high intracranial pressure/brain herniation, status epilepticus (convulsive and nonconvulsive), acute ischemic stroke, intracranial hemorrhage, and cerebral venous sinus thrombosis. We also briefly discuss the effects of SARS-CoV-2 on the CNS. Many patients present with more than one problem and more than one cause. This multiplicity can make diagnosis and management difficult. The goal is to preserve life and function, achieved by early, correct diagnosis. Still, patients come late in the disease course, or these neurological emergencies unfold quickly and catastrophically. Acute mental status change is the top neurological symptom in cancer patients; it is the final common path to intracranial hypertension, epileptic seizures, infection, electrolyte abnormalities, intracranial hemorrhage, ischemic stroke, and adverse effects from drugs. The emergency physician should also be comfortable in starting discussions on goals of care and advance care planning before admission to other hospital areas.

P. Rai

A. Shivaprasad Stanley H. Appel Department of Neurology, Houston Methodist Hospital, Houston, TX, USA

I. W. Tremont-Lukats (⊠)

Kenneth R. Peak Brain and Pituitary Tumor Center, Department of Neurosurgery, Houston Methodist Hospital, Houston, TX, USA e-mail: itremont@houstonmethodist.org

S. Tummala

Approach to the Patient with Altered Mental Status

Mental status is a state of wakefulness in alert persons. Changes in mental status can also affect cognition, including orientation to self, place, and time; the ability to register and recall objects (memory registration and recall after minutes); and the ability to understand and execute more complex commands, such as backward spelling or arithmetic operations (attention/concentration) and language.

Changes in mental status can occur in a patient with normal mental function or can appear against the backdrop of a chronic cognitive disorder such as dementia or mental retardation. In the medical literature, the terms altered mental status, changes in mental status, confusion, encephalopathy, and delirium are often used interchangeably. Encephalopathy and delirium are more relevant as diagnostic keywords. For simplicity, we will use delirium in this chapter for any acute change in arousal and mental function.

Frequency and Importance

In cancer, delirium is a life-threatening complication and a powerful driver of emergency services. Care providers in the ED frequently overlook delirium, with a reported misdiagnosis rate of 41% in the ED of an NCI-designated comprehensive cancer center [1, 2]. Its frequency ranges from 57 to 85% in cancer patients compared to 15–30% in medically ill hospitalized patients [3]. Among 352 cancer patients in the acute palliative care unit at The University of Texas MD Anderson Cancer Center, delirium was the most common diagnosis in 43% at admission, increasing to 70% during the entire stay [4]. At Memorial Sloan Kettering Cancer Center (MSKCC), delirium accounted for 27% of all inpatient neurology consults. This figure did not include psychiatry consults or unrecognized cases [3]. A review of 771 inpatient palliative care consults found that referring teams missed the



Department of Neurology, University of Texas Medical Branch, Galveston, TX, USA

Department of Neuro-Oncology, MD Anderson Cancer Center, Houston, TX, USA

diagnosis in 61% of cases, of which 63% had hypoactive delirium [2].

Causes

Many patients with cancer have more than one reason for delirium [5, 6]. The most important causes in the ED are illustrated in Fig. 16.1. The first three suspects to investigate are drugs, sepsis, and organ failure [6]. Opioids and benzodiazepines are the most common drugs causing delirium in the ED, in surgical recovery rooms, and during hospitalization. It is impossible to quantify the relative weight of each drug in cases of polypharmacy, especially when accompanied by renal or liver impairment and advanced age [7].

The worldwide emergence of coronavirus disease (COVID-19) due to the severe acute respiratory syndrome coronavirus 2 (SARS-CoV-2) is associated with higher mortality rates in cancer patients [8, 9]. A surveillance study in the UK reported AMS as the second most common neurologic complication in COVID-19, in 31% of 125 cases [10]. We currently do not have longitudinal data mature enough to estimate the frequency, course, and prognosis of delirium from COVID-19 in cancer, or whether it is from neuroinvasion, hypoxia, or both [11, 12]. The current view is that the COVID-19-related encephalopathy is the result of age, comorbidities, and the systemic inflammatory response to the infection, including adult respiratory distress syndrome (ARDS) resulting in cerebral hypoxia in postmortem exami-



Fig. 16.1 Causes of abnormal mental status of patients in the emergency department. *BZD* Benzodiazepines; *ED* Emergency Department

nation [12, 13]. Preliminary evidence in non-cancer patients admitted to intensive care units with ARDS showed that, in addition to other neurologic findings, delirium was frequent (65%). Approximately 33% of survivors remained disoriented and inattentive after discharge [14].

A long list of chemotherapy agents, immune checkpoint inhibitors (ICI), and chimeric antigen receptor (CAR) T-cell therapies can also cause central neurotoxicity (Table 16.1) [15–17]. The immune-effector cell-associated neurological syndrome (ICANS) is a distinct, common complication of adoptive T-cell immunotherapy [18]. Some of these drugs have been linked to the posterior reversible encephalopathy syndrome (PRES). Physicians caring for cancer patients in the ED should consider PRES if symptoms and findings (confusion, headaches, cortical blindness, seizures, and hypertension) suggest the diagnosis and patients are on chemotherapy or immunosuppressive drugs [19, 20]. Another cause is progressive multifocal leukoencephalopathy (PML) among leukemia and lymphoma patients treated with monoclonal antibodies targeting B and T cells. Delirium is also one of the complications of allogeneic HSCT due to chronic immunosuppression for the prevention of graft versus host disease [21].

Diagnosis

Two elements are essential to the diagnosis of delirium: (1) acute or subacute onset in minutes, hours, or even a few days and (2) fluctuation in attention levels.

There are several instruments for the assessment of mental status in cancer patients [56]. The mini-mental state examination (MMSE) is easy to learn and use. It can detect encephalopathy within 1 minute, especially when assessing orientation, recall, and attention/concentration [57]. The language domain is generally unremarkable in mild or moderate delirium. The MMSE is useful to evaluate and diagnose delirium in patients with baseline dementia, but it is not adequate to screen for dementia itself. In the ED, pre-existing dementia is often based on information on prior cognitive function from relatives or friends.

Assessment and Management

Not every patient needs all the tests in Fig. 16.2. The essential laboratory data and procedures for adequate diagnosis of encephalopathy and its cause(s) are contained in the two upper bullets of the diagram. The treatment of delirium is supportive, and its resolution depends on control of the underlying problems; if these complications resolve, encephalopathy should improve. Figure 16.3 depicts the principles of treatment, as reviewed in detail elsewhere [7].

Table 16.1 Ch	hemotherapy and	other anticancer of	drugs associated	with enceph	nalopathy/delirium
---------------	-----------------	---------------------	------------------	-------------	--------------------

Drug	Frequency	Comments
Bevacizumab	Unknown	Associated as monotherapy, or with other drugs in PRES [19, 22, 23]
Bortezomib and carfilzomib	Unknown	Well-known association with PRES [24, 25]
Blinatumomab	All grades: 36–53% ≥ grade 3: 7–14%	Pooled data [26, 27]
CAR T-cell therapy: Axicabtagene ciloleucel (YESCARTA®) Tisagenlecleucel (Kymriah®) Brexucabtagene autoleucel (Tecartus®)	ICANS (any grade): 68% [28] ≥ grade 3: 30–41% [28–30]	Stereotypical presentation, biphasic pattern. Early: tremor, dysgraphia, aphasia. Aphasia can be prominent later and is very specific for this class [18]. PRES in 2% [28]
Cisplatin	Unknown but rare	Described with PRES
Etoposide	Unknown but rare	Described with PRES
Fludarabine	Very rare but serious	
5-Fluorouracil	Very rare	Check for dihydropyrimidine dehydrogenase (DPD) deficiency
Gemcitabine		
Ifosfamide	15–21% [31, 32]	Risk is higher in ECOG PS 2–4, and increase in serum creatinine levels [33]
Interferons (α and β)	Rare, discrepant reports on severity [34, 35]	No association with PRES
Interleukin-2	Rare [36–38]	
Ipilimumab	<1% [39]	Described with PRES [40]
L-Asparaginase	Unknown, case reports [41]	Mostly during consolidation associated with PRES [42]
Methotrexate	With sporadic reports since the 1970s, frequency unknown	Described with parenteral [43–45] and intraventricular administration [46–48] associated with PRES [49]
Nitrosoureas	Unknown	Causality is difficult to demonstrate because the current use of these drugs is to treat CNS tumors
Procarbazine	Unknown; association is very weak	Described with CCNU and vincristine [50], not in PRES
Sunitinib	Unknown	Associated with PRES [51]
Tamoxifen	No firm association [52]	Not described in PRES
Thiotepa	18% with high-dose regimens [53]	Use of tramadol may increase the risk of neurotoxicity
Vincristine	Unknown	Causality unclear, associated with PRES [54, 55]

CAR T Chimeric antigen receptor T-cell therapy, PRES posterior reversible encephalopathy syndrome, ICANS immune-effector cell-associated neurological syndrome, ECOG Eastern Cooperative Oncologic Group, PS performance status



BZD: Benzodiazepines

Fig. 16.2 Diagnostic sequence in the diagnosis of altered mental status (ABG arterial blood gases, EEG electroencephalogram, LP lumbar puncture)

Brain Herniation

Principles

Intracranial volume is the sum of the volumes of brain tissue, cerebrospinal fluid (CSF), and blood, within a rigid compartment (skull) [58]. This means that changes in relative volumes will not alter the total intracranial volume (Monro-Kellie doctrine). Figure 16.4 illustrates the interplay among these compartments and how an increase in intracranial pressure (ICP) can lead to brain herniation. High ICP triggers a compensatory shift of CSF into the spinal subarachnoid space and a reduction of blood volume by the cerebral venous system. If the cause of high ICP supersedes these homeostatic mechanisms, ICP will rise, and the brain parenchyma can herniate in three directions following a gradient pressure: under the falx (subfalcine herniation); past the tentorium cerebellum (uncal or transtentorial herniation); and past the foramen magnum (tonsillar herniation) [59]. Early recognition of impending



Fig. 16.3 Emergency department management of altered mental status in cancer patients

brain herniation will allow prompt treatment (Fig. 16.5). Miller Fisher argued that the crucial and irreversible injury are the events preceding herniation, such as displacement of midline structures against the tentorium or overcrowding of the posterior fossa [60], and this is supported by clinical observations [61].

Causes

In oncologic practice, cerebral edema and an intracranial mass are the most common causes of brain herniation syndromes. The intracranial mass can be (1) a primary tumor (e.g., glioblastoma, meningioma with brain invasion, or ependymomas); (2) a brain metastasis (single or multiple); and (3) radiation necrosis. In these cases, the perilesional edema is from BBB disruption (vasogenic) and can respond to corticosteroids.

Alternatively, nontumoral lesions with focal mass effects may cause intracellular (cytotoxic) edema unresponsive to steroids. Causes include (1) hypertensive hematomas; (2) traumatic or non-traumatic intracranial hemorrhage; (3) ischemic infarcts in the distribution of a large vessel such as the carotid artery or main trunk of the middle cerebral artery (MCA); and (4) vasculitis (if ischemia or hemorrhage predominates and the vasculitis is not due to an autoimmune process).

A third group has high ICP with diffuse cerebral cytotoxic edema from widespread cellular injury and includes the following: (1) hypoxia from cardiorespiratory arrest; (2) refractory convulsive epileptic status; (3) liver or renal failure; and (4) hydrocephalus (in neoplastic meningitis).



16 Central Nervous System



Fig. 16.5 Patient with metastatic B-cell lymphoma. (**a**) T1-gadolinium axial MRI. The left uncus is shifting against the cerebral peduncle (impending uncal herniation). (**b**) MRI of the brain, T1 with gadolin-

ium, coronal section. Another view of the uncal displacement (*straight arrow*) and a simultaneous subfalcine herniation

Assessment

Not all patients with intracranial hypertension present with the same symptoms and findings; we have seen patients with radiographic evidence of impending herniation but a normal neurological examination, although they are rare (<5%) [62]. Many others do not have herniation but a mass effect, neurological deficits, and AMS. In general, headache, neck pain, AMS, or seizures can indicate high ICP. Papilledema is a useful sign with low sensitivity, and very few physicians in the ED have funduscopy skills to identify papilledema (Fig. 16.6). Point of care ultrasound may be a useful adjunct to fundoscopy in making the diagnosis of intracranial hypertension [63].

Other patients can present with acute bursts of neurologic dysfunction, often mistaken for epileptic seizures; many of them have mass lesions or hydrocephalus from leptomeningeal involvement. Such episodes are due to a sudden rise in intracranial pressure (plateau waves) and last from 1 to 20 min (Table 16.2). The EEG can sometimes show indirect clues of increased ICP with frontal intermittent rhythmic delta activity (FIRDA) patterns (Fig. 16.7).

Management and Prognosis

After an ED diagnosis of high ICP with or without evidence (clinical or radiographic) of brain herniation, the



Fig. 16.6 Papilledema in a 53-year-old woman in the ED with new onset, daily headaches. Bedside fundoscopy quickly established a secondary headache. Opening cerebrospinal fluid pressure: 40 cm H20. Cytology: adenocarcinoma cells. Blurred optic disc contours, blurred retinal vessel walls, and soft exudates

specific circumstance will dictate the urgency of treatment. The choice of medical interventions depends on the cause responsible for high ICP, and these include hyperventilation, osmotherapy, and corticosteroids (Table 16.3). Hyperventilation can be started in the ED, with advanced airway management and mechanical ventilation before transfer to intensive care. Mannitol and corticosteroids can begin in the ED and continue on the hospital floor. Most hypertonic saline infusions, particularly 7.5%, 15%, and 23.4% sodium chloride, require ICU monitoring. The prognosis of symptomatic, acute brain herniation in cancer patients is dismal, with most dying in minutes or hours regardless of the cause. Even if they survive resuscitation efforts, they will remain unresponsive and comatose until death. Patients with metastatic or primary brain tumors and gradual elevation of

 Table 16.2
 Signs and symptoms associated with plateau waves [5]

Altered	consciousness	(delirium	etupor	or coma)
Alleled	CONSCIOUSNESS		SILLINOI	OI COIDAL

"Spells" of blank stare (confused with partial complex seizures) Spontaneous, wide oscillations of blood pressure, respiration, or heart rate

Headache, pain in neck or shoulders
Nasal pruritus
Nausea/vomiting
Facial flushing
Shivering, goosebumps, sweating
Temperature increase
Yawning or hiccups
Opisthotonus
Mydriasis
Weakness of cranial nerve III or VI
Nuchal rigidity
Clonic movements of extremities
Decorticate or decerebrate posturing
Bilateral extensor plantar signs

These episodes are paroxysmal, may last minutes, and may be triggered by touch, pain, suction, positioning, or noise

ICP are more likely to respond to osmotherapy, steroids, and debulking surgery, if indicated.

Status Epilepticus (SE)

Definition and Classification

Status epilepticus can be convulsive (CSE) or nonconvulsive (NCSE), and patients can evolve from one to another in the ED. CSE is easily recognizable, with partial (focal) tonicclonic, generalized tonic-clonic, or predominant tonic posturing or clonic movements (rhythmic jerking), or mixed. Convulsive status epilepticus (CSE) is a life-threatening emergency [65].

Status epilepticus from some regions of the brain (temporal lobe, occipital and deep frontal) might not have overt motor manifestations and typify a complex partial status epilepticus. These events require observation of subtle movements or behavioral changes by observant witnesses or care providers:

- · Licking behavior
- Lip-smacking
- · Chewing, tooth grinding, spitting
- Grunting or guttural sounds
- Aversive or versive eye/head movements
- Nystagmus
- · Myoclonus of eyelids, face, or perioral regions
- Picking/fumbling, facial grimacing, gesticulating activity
- Isolated focal facial, finger, or toe twitching



Fig. 16.7 Frontal dysfunction with delta (FIRDA, boxes). Opening cerebrospinal fluid pressure: 27 cm H2O. Patient with plateau waves

Intervention	Dose	Onset/duration	Advantage(s)	Disadvantage(s)
Hyperventilation	Lower pCO2 to 25–30 mmHg by increasing respiratory rate with same tidal volume	Seconds/minutes	Fastest onset Effective for high ICP regardless of the cause	Very short duration, with effect lost after several hours Will require endotracheal intubation May be more harmful to an already injured brain
Osmotherapy (mannitol, 20% solution)	1 g/kg iv initial dose; next dose: 0.5 g/kg and 0.25 g/ kg q6h	Within 15–20 min; max effect is in 1 h Keep osmolality between 210 and 320 mOsm/L	Longer duration Effective regardless of the cause Note: still is the standard of care in cancer patients	Rebound effect is possible Hyperosmolality and acute renal failure
Osmotherapy (hypertonic saline solutions)	3%: 250 mL 7.5%: 250 mL 23.4%: 30–60 mL	Onset: within min	Effective when mannitol is not Needs central line for administration – could be considered for rapidly evolving herniation	Minimal experience in brain tumors Safer than mannitol but it may have adverse effects: CHF, hyperchloremic acidemia, hypernatremia, seizures
Corticosteroids (dexamethasone most used)	Initial dose: 10 mg, then 4–6 mg iv q4–6 h. Decrease as soon as possible	24–36 h/days	Reliable and steady effect in vasogenic edema from metastatic or primary brain tumors	Not effective in anoxic-ischemic or toxic edema AE: hyperglycemia, mood changes, insomnia, immunosuppression, Cushing syndrome, skin frailty, accelerated bone resorption Myopathy can be disabling

Table 16.3 Emergency medical treatment of patients with de facto or impending cerebral herniation [28]

ICP Intracranial pressure; CHF congestive heart failure; AE adverse effect

- Pelvic thrusting and bicycling movements
- Asymmetric tonic posturing ("sign of four")
- Dystonic limb posturing

These symptoms and signs require bedside EEG for prompt diagnosis. NCSE also presents with AMS and behavior changes from baseline, without overt motor manifestations. The list is diverse and includes:

- Altered consciousness
- Impairment of verbal skills
- · Impairment of clock drawing
- Indifferent attitude
- Fear
- Rare hysteria
- Suicidal thoughts
- Hallucinations

Bedside EEG is essential for prompt diagnosis. Status epilepticus involves ongoing or recurrent seizures *without* recovery of sensory, motor, or cognitive function or consciousness to baseline. The duration of seizures in the definition of CSE has changed over time from 30 min to 5 min; we consider seizures lasting 2 minutes or longer in cancer patients as an emergency requiring prompt treatment and workup. There is no consensus on how soon to intervene; other experts recommend treatment after 5 min [64]. Unlike CSE, NCSE requires timely recognition without overly aggressive treatments. NCSE has a prevalence of 15% among critically ill cancer patients and seems associated with lower mortality, normal kidney function, and hematologic malignancies receiving chemotherapy [66]. Postictal deficits (Todd paralysis) include temporary limb weakness or hemiplegia after a seizure, with new or worsening of prior known weakness, and clinically masquerade as an ongoing seizure. The EEG is necessary to identify surreptitious seizure activity.

Causes

Epileptic seizures are a frequent symptom of CNS metastases (leptomeningeal, dural, or parenchymal), primary brain tumors (meningiomas, astrocytomas, and oligodendrogliomas, with oligodendrogliomas having the highest seizure rate), metabolic disorders (hyponatremia, hypoglycemia, hypoxia, and hypercalcemia), CNS infections (herpesvirus), intracranial hemorrhage (spontaneous or traumatic), ischemic infarctions, and treatment-related factors, as well as paraneoplastic limbic encephalitis.

In the era of immunotherapy, we increasingly recognize autoimmune encephalitis, meningitis, and demyelinating brain lesions. We also see seizures associated with blinatumomab and CAR T-cell therapy. If these patients return to the ED with AMS, it is necessary to perform a complete workup (sepsis workup, MR brain, EEG, lumbar puncture).



Fig. 16.8 (a) Nonconvulsive convulsive status epilepticus with ifosfamide toxicity: disorganized background with generalized and multifocal periodic complexes that are isolated (*arrow*) and at 2.5 Hz (*box*).

(**b**) After 1.5 mg lorazepam: organized background (*box*) with attenuation of the periodic complexes (*arrow*). Patient able to state her name and follow one-step commands. Normal MR brain

Assessment

ED providers should observe or inquire for behavior changes, subtle movements, and AMS. Manifestations of NCSE can range from being awake, able to count from 1 to 20 but not backward, to confusion and overt coma. Clinical and EEG improvement to a benzodiazepine challenge (Fig. 16.8) can facilitate the diagnosis. The EEG has limitations, especially with intermittent seizures. Prolonged EEG or repeat EEG can increase the detection level, and a

nondiagnostic EEG does *not* rule out a seizure. Clinical judgment should prevail, acknowledging the limits of diagnostic tests, and with high suspicion it is clinically and ethically justified to start benzodiazepines and anticonvulsant drugs [67], and we recommend upfront treatment if EEG is not readily available. An EEG should be performed following resolution of CSE if the patient's mental status and focal neurological deficits have not recovered to baseline. The aim is to rule out subclinical electrographic seizures.

Management and Prognosis

The initial assessment and treatment of CSE is detailed in Fig. 16.9. After securing ventilatory and circulatory support, the early and rapid use of benzodiazepines, phenytoin, levetiracetam, lacosamide, sodium valproate, phenobarbital, propofol, midazolam, and, in some cases, pentobarbital may prevent progression to refractory and super-refractory status epilepticus. Any seizure longer than 2 minutes should be treated aggressively. If the duration is uncertain, treatment has to be equally aggressive, even though the probability of success ("breaking" the SE) is less if the patient has been in status for hours or days. The prognosis is complicated and depends on the duration of status, underlying causes, and presence of additional complications. CSE and partial status epilepticus can contribute to further neurological injury in cancer patients due to cytotoxic edema and intracranial hypertension.

NCSE generally requires 1 to 4 mg lorazepam in different doses and sequences, for example, 1 mg doses as needed with reassessments in between or 2 mg plus additional doses as needed. Drugs with little or no sedating effects (levetiracetam or lacosamide) are preferred. Following the resolution of NCSE, we recommend lowerdose intermittent lorazepam 0.5 mg iv q4–8 hrs x 24 hrs. Follow-up EEGs can often be indeterminate with trials of benzodiazepine and antiepileptic drugs.

NCSE and CSE share similar etiologies, and it is essential to address the root cause quickly. In our experience, leptomeningeal disease, drugs (ifosfamide, cephalosporins), and acute or acute-on-chronic renal injury are among the most common reasons. For both conditions, clinical improvement and EEG (Fig. 16.8b) are therapeutic endpoints.

Intracranial Hemorrhage (ICH)

Importance and Causes

Although the overall incidence is unknown, ICH is a frequent ED diagnosis in any institution caring for patients with cancer. ICH is a generic term that includes parenchymal, subarachnoid, subdural, and epidural hemorrhage. Although non-traumatic, ICH can occur as an adverse effect of supratherapeutic anticoagulation (but not therapeutic anticoagulation for venous thromboembolic disease [68]) or due to profound or prolonged



Fig. 16.9 Management guidelines for convulsive status epilepticus in cancer patients at the University of Texas MD Anderson Cancer Center. MAP Mean arterial pressure, POC point-of-care, CBC complete blood count, ABG arterial blood gases, AED antiepileptic drugs, LMD lepto-

meningeal disease, WBXRT whole brain radiation therapy, SRT stereotactic radiotherapy, ABX antibiotics, IVIG intravenous immunoglobulin

thrombocytopenia caused by drugs, disseminated intravascular coagulation (DIC) [69], or sepsis.

In solid tumors, bleeding can be spontaneous because of neovascularization (melanoma, lung cancer, renal cell carcinoma, or glioblastoma). ICH is an adverse effect with bevacizumab, but there is no evidence of additional risk in these patients [70]. The 72 h mortality rate of ICH in leukemia ranges between 12% and 19%. After 30 days, it climbs up to 40% [71]. Old age and high disease burden increase the risk of death [69].

Presentation

Patients with ICH can present with symptoms ranging from an incidental finding on CT scan of the brain to the classic picture of sudden headache, AMS (including stupor and coma), focal deficits, and epileptic seizures. Incidental or mildly asymptomatic presentations are more frequent in patients with intratumoral bleeding. Patients without a history of cancer may present with an ICH as the initial manifestation of neoplasm.



Fig. 16.10 Intratumoral hemorrhage in glioblastoma. Axial T2* MRI of the brain of a 49-year-old man presenting with a 7-day history of confusion. Fluent (Wernicke-type) aphasia on exam. No headaches

Diagnosis

A CT scan confirms ICH in most cases. In doubt, the MR brain with T1 with/without gadolinium and gradient echo or T2* (star) sequences are most helpful (Fig. 16.10). Bleeding in venous angiomas, cavernomas, incidental aneurysms, and elderly patients with amyloid angiopathy can sometimes pose a problem of differential diagnosis. In most cases, a careful analysis of the MRI sequences correctly identifies the cause.

Treatment and Prognosis (Fig. 16.11)

The treatment of ICH depends on the location of bleeding and cancer status. For prognosis, see section "Importance and Causes" above. Aggressive platelet and factor transfusion is indicated with the goals to keep fibrinogen >150 mg/ dl and platelets >50 × 10⁹ L⁻¹. Most patients with solid tumors and ICH that are symptomatic but stable can be managed conservatively. Critically ill patients usually arrive after cardiac arrest or are stuporous or comatose. These patients, if they survive, have severe and irreversible neurological deficits that will exclude them from further oncologic interventions, and supportive care is the best option.

Acute Ischemic Stroke

Importance and Causes

Patients with cancer have a high risk of arterial cerebrovascular ischemia, twice as high as the general population [72, 73]. The risk is highest before primary tumor treatment, in gastrointestinal adenocarcinomas, in brain tumors, in patients who have had neck irradiation to treat head/neck neoplasms, and in advanced stages [74].

In a large population-based study, brain and GI tract tumors (colorectal, pancreas, and liver) had the highest mortality rates; stroke clustered in patients treated for brain



Fig. 16.11 Treatment alternatives in intracranial hemorrhage





Fig. 16.12 (a) Axial, diffusion-weighted imaging (DWI) sequence. Multiple, bilateral embolic infarcts visible in cerebellum and supratentorial areas of a 65-year-old woman with lung adenocarcinoma. Transesophageal echocardiogram showed aortic valve leaflet with thrombus in the right atrium. (b) Multifocal cerebral and cerebellar diffusion restriction signals in DWI/T2 sequences. Cortical infarcts in

multiple vessel distributions in a 26-year-old patient with sarcoma. MRA/MRV/2D echocardiograms were normal. These cortical strokes are due to meningeal pial penetrating arteries infiltrated by carcinomatosis meningitis or bacteria. In this case, bacterial meningitis. Source: an intravenous pain pump

tumors and lymphomas if younger than 40 years and in prostate, breast, and colon if older than 40 years of age [73]. The classical, conventional stroke risk factors so prevalent in the US population compound the risk even more: age; hypertension; atrial fibrillation; obesity; diabetes; and hypercholesterolemia. Other powerful stroke triggers in cancer patients, ranked from more to less common [5, 74], include (1) hypercoagulability and disseminated intravascular coagulation (DIC); (2) nonbacterial thrombotic endocarditis (NBTE) (Fig. 16.12a); (3) septic emboli (bacterial or fungal) (Fig. 16.12b); (4) tumor emboli; and (5) accelerated atherosclerosis after radiation therapy (Fig. 16.13).

Diagnosis

Most cancer patients in the ED present with three well-defined stroke syndromes [74, 75]. Those with *focal neurological deficits* (e.g., aphasia, hemiparesis, hemi-inattention, or homonymous temporal hemianopsia) are the most straightforward. Postictal deficit (Todd paralysis) should be included in the differential diagnosis, as it is a notorious stroke mimic among cancer patients. *Delirium* without focal signs is a more significant diagnostic challenge. Encephalopathic patients may present without lateral signs (focal deficits). ED providers *must* include ischemic stroke in the differential diagnosis of patients with delirium and no other apparent neurological deficits. Multiple, bilateral embolic infarcts in the carotid and vertebrobasilar territories are common in this group [74, 76]. Finally, patients may present with *both focal deficits and delirium*.

Assessment, Treatment, and Prognosis

The diagnostic sequence is similar to the diagnosis of stroke in the general population. An initial CT head without contrast screens for hemorrhage. The best tool to confirm the diagnosis is brain MR with diffusion-weighted imaging (DWI). Apparent diffusion coefficient (ADC) values may indicate areas of ischemic penumbra, and if there is no description of ADC changes in the radiology report, the emergency physician should ask about them. Fluidattenuated inversion recovery (FLAIR) will help determine the approximate age of the ischemic lesion, and T1 without gadolinium and gradient echo (GRE) will assist in the diagnosis of hemorrhagic transformation from reperfusion injury, a spontaneous event in ischemic strokes or after thromboly-



Fig. 16.13 47-year-old woman with history of left oral cancer, treated with radiation and chemotherapy 7 years before. She came to the ED with acute onset aphasia, right hemiparesis, and fall. National Institutes of Health (NIH) Stroke Scale score: 20. CT of head (*left panel, arrow*): classic hyperdense middle cerebral artery (MCA) sign from acute

thromboembolic occlusion. Also: loss of left insular ribbon from cytotoxic edema, suggesting onset longer than 6 h (*yellow star, compared to blue star on the right*). *Right panel:* CT angiogram. Thrombus in the left internal carotid artery

sis. Finally, an MR angiogram of head and neck and MR venogram can assess for extra- or intracranial stenosis, vasculitis, and venous or sinus thrombosis.

The treatment of acute stroke is conditional to the status of underlying cancer or intercurrent complications derived from the tumor or its treatment. It is obviously more straightforward to treat patients in remission and not on active therapy than patients with complications of chemotherapy, sepsis, or active DIC.

In principle, all patients should have control of hypertension, diabetes, and hypercholesterolemia. If no contraindications exist, aspirin 325 mg daily for secondary prevention and oral anticoagulation for patients with atrial fibrillation are appropriate. Aspirin 100 mg daily for primary prevention of cardiovascular or cerebrovascular events in cancer patients has no evidence for widespread use yet and requires a thoughtful benefit-harm discussion based on patient preference [77]. Patients within 4.5 h of stroke onset are candidates for iv thrombolytics, as long as there is no absolute contraindication [78, 79]. A brain tumor is not in itself a contraindication, but a recent hemorrhagic metastasis poses additional risk, and most patients arrive past the therapeutic window, are too unstable, or have contraindications to thrombolytic therapy. Only 1.6% of patients with cancer and acute stroke receive iv thrombolytics or endovascular therapy [78].

The prognosis of acute ischemic stroke in patients with cancer is generally poor [80]. In patients after thrombolytics,

the 3-month mortality is about 50%, and among survivors, 50% had a poor functional outcome. Patients with solid tumors and metastases have worse outcomes than patients with hematologic malignancies. In those undergoing antineoplastic therapy, a stroke resulting in worse functional outcome means poor performance. This consequence has relevance for goals of care, quality of life, and eligibility for clinical trials, among other issues.

Cerebral Venous Sinus Thrombosis

Frequency

Venous sinus thrombosis occurs in 9% of adults with hematologic malignancies or solid tumors. It affects more women than men and tends to happen within the first year of cancer diagnosis [81].

Presentation

Headache, confusion, and seizures are common presentations. As in other emergencies, the severity of symptoms varies from an incidental finding to coma. Younger patients tend to present with headache and papilledema (increased ICP), whereas older patients (>50 years) have more AMS [82].

Diagnosis

A venogram by CT or MRI is the most direct means of noninvasive diagnosis [83]. The regular MRI of the brain can point to the suspect area before the angiogram.

Treatment and Prognosis

If asymptomatic, we tend to treat with anticoagulation, with admittedly little evidence that this approach is better than observation. For symptomatic, stable patients, supportive measures are the first steps, followed by anticoagulation with subcutaneous low-molecular-weight heparin (LMWH). The optimal duration is unclear, but many patients receive treatment for 3 to 6 months, and some permanently, reflecting how little we know and the need for well-controlled studies. Thrombectomy and thrombolysis with the tissue plasminogen activator (tPA) are effective interventions in selected cases. We have no experience with these agents as most of our patients have contraindications, or their prognosis is poor. For stable and symptomatic patients, the prognosis for survival and neurologic function is good. For patients with an acute presentation with brain herniation and CSE, the outlook is poor with a high mortality rate. Predictably, the prognosis for survival or recovery in cancer patients is worse than in non-cancer patients [75].

Case Studies

Case Study 1: Pembrolizumab-Associated Myopathy

A 73-year-old man with renal cell carcinoma, bone, and lung metastases began treatment with pembrolizumab (ICI) every 21 days and axitinib (a tyrosine kinase inhibitor). After cycle 4, patient came to the ED with diffuse bilateral muscle tenderness and weakness, ascending to involve neck and bulbar muscles. On examination, he exhibited severe neck ptosis (antecollis), eye ptosis, dysarthria, and dysphagia (requiring a feeding tube). He had acceptable respiratory function, without need for supplemental oxygen. Before arrival, he had received IVIG and high-dose prednisolone at another institution with partial improvement.

Our initial clinical diagnosis was acute inflammatory myopathy. There is a clear association between ICI, myositis, and myasthenia gravis. His CK levels rose from 49 U/L on admission to 545 U/L on day 19 (72 h after admission) and 343 U/L on day 21. EMG was consistent with myositis in the cervical paraspinals and trapezii. A muscle biopsy was nondiagnostic twice and anti-titin and striatal antibodies were negative. He was treated with plasma exchange and rituximab 375 mg/m2 BSA days 1 and 15, plus infliximab 5 mg/kg days 1 and 15. Steroids were tapered over 6 weeks. Mild neurological disability persisted, and although the patient is ambulatory, he still speaks in a low voice and has slurred speech and dysphagia for solids.

Case Study 2: Atezolizumab-Associated Meningoencephalitis [50]

A 53-year-old woman with metastatic squamous cell carcinoma of the cervix was referred to the ED by her gynecologic oncologist for evaluation of headache and AMS for 24 h. Thirteen days earlier she had received her first doses of atezolizumab and bevacizumab.

She was found to be delirious, with nuchal rigidity. Laboratory evaluation revealed a normal absolute neutrophil count (ANC), lymphopenia, and a white blood cell count of 3.4 10⁹/L. Brain MRI suggested diffuse leptomeningeal enhancement along the cerebral sulci and cerebellar foliae (Fig. 16.14). Supportive care was initiated in the ED. She was transferred to the ICU and treated with dexamethasone (Fig. 16.15).

On day 6 after admission, her mental status deteriorated. A routine EEG showed a pattern of nonconvulsive status that responded readily to lorazepam (Fig. 16.16). After this event, the patient steadily recovered, and corticosteroids were tapered after 21 days of therapy. She had no residual neurological disability and became fully independent.

Case Study 3: CAR T-Cell Neurotoxicity

A 58-year-old man with diffuse large B-cell lymphoma was in relapse after multiple lines of treatment. He had received axicabtagene ciloleucel (YESCARTA®) on day 0 and on day 3, and he exhibited cytokine release syndrome (CRS) grade 2 with fever, fatigue, nausea, and malaise. He was managed accordingly, with symptomatic treatment and observation in the fast-track clinic as an outpatient. This mild CRS was resolved by day 8 when his family brought him to the ED for acute onset confusion. He was alternating between moments of combativeness with sleepiness, and this was a departure from his normal behavior.

On examination the emergency physician found that patient was disoriented, with no attention span, and mostly uncooperative. He could not speak in complete sentences, and could not write when his relative gave him a paper and pen to help him communicate. There was no evidence of concurrent CRS, and the ED physician diagnosed neurotoxicity grade 2 related to YESCARTA®. He was treated immediately with dexamethasone 10 mg iv q6h and admitted. He also began levetiracetam, a bolus of 20 mg/kg and maintenance dose of 1000 mg iv q12h. Besides sinus tachycardia, all vitals were normal, and laboratory values, including liver and



Fig. 16.14 *Case study 2*: (a) axial MR brain, T1 with contrast, leptomeningeal enhancement (*arrows*) in case study 2, on day 2 of symptom onset. (b) At day 8, after high-dose iv steroids

	Dexamethasone 4mg/day ↑24 mg/day	Opens eyes, follows commands	Dexamethasone tapered over 21 days
CSF (units)	Day 0	Day 7	Day 15
WBC (10 ⁹ /L)	553 (H)	9 (H)	3
Protein (g/L)	>600 (H)	302.0 (H)	71.0 (H)
Beta 2 Microglobulin (mcg/mL)		10.0 (H) 0.7 - 1.8 mg/L	
MBP (ng/mL)		21.30 (H) 5.50 ng/mL	
Albumin (mg/dL)		119.0 (H)	22.7
Cytology	Numerous acute inflammatory cells		-

Fig. 16.15 *Case study 2*: Treatment course, with serial cerebrospinal fluid (CSF) findings revealing high CSF white blood cells, protein, beta-2-microglobulin, and basic myelin protein levels



Fig. 16.16 *Case study 2:* acute altered mental status on day 6, with EEG diagnostic of nonconvulsive status before (*upper panel*) and after treatment with antiepileptic drugs (*lower panel*)

renal parameters, were normal. Urine analysis and chest *x*-ray were unremarkable.

The patient went to the regular floor where dexamethasone continued for 4 more days (day 12). He improved steadily within 72 h and was at baseline by day 14. His treating team stopped dexamethasone and the patient was discharged.

Comment: In the trials leading to the FDA approval of YESCARTA®, 85% of patients had neurotoxicity, 31% of them grade 3 or higher. Most cases (98%) happened within 8 weeks of the infusion day (day 0), with a median onset at 4 days (range: 1–43 days). Our patient began neurological symptoms on day 8. The median duration of neurotoxicity is 15–17 days, with rare cases having encephalopathy for almost 6 months. For grade 2 toxicity and no concurrent CRS, dexamethasone, AED, and hemodynamic monitoring in a regular floor are appropriate.

References

- Elsayem AF, Bruera E, Valentine A, Warneke CL, Wood GL, Yeung SJ, et al. Advance directives, hospitalization, and survival among advanced cancer patients with delirium presenting to the emergency department: a prospective study. Oncologist. 2017;22(11):1368–73.
- de la Cruz M, Fan J, Yennu S, Tanco K, Shin S, Wu J, et al. The frequency of missed delirium in patients referred to palliative care in a comprehensive cancer center. Support Care Cancer. 2015;23(8):2427–33.
- Nolan C, DeAngelis LM. The confused oncologic patient: a rational clinical approach. Curr Opin Neurol. 2016;29(6):789–96.
- Hui D, dos Santos R, Reddy S, Nascimento MS, Zhukovsky DS, Paiva CE, et al. Acute symptomatic complications among patients with advanced cancer admitted to acute palliative care units: a prospective observational study. Palliat Med. 2015;29(9):826–33.
- DeAngelis LM, Posner JB. Neurologic complications of cancer. 2nd ed. Oxford/New York: Oxford University Press; 2009.
- Tuma R, DeAngelis LM. Altered mental status in patients with cancer. Arch Neurol. 2000;57(12):1727–31.

P. Rai et al.

- Lawlor PG. Cancer patients with delirium in the emergency department: a frequent and distressing problem that calls for better assessment. Cancer. 2016;122(18):2783–6.
- Assaad S, Avrillon V, Fournier ML, Mastroianni B, Russias B, Swalduz A, et al. High mortality rate in cancer patients with symptoms of COVID-19 with or without detectable SARS-COV-2 on RT-PCR. Eur J Cancer. 2020;135:251–9.
- Mehta V, Goel S, Kabarriti R, Cole D, Goldfinger M, Acuna-Villaorduna A, et al. Case fatality rate of cancer patients with COVID-19 in a New York hospital system. Cancer Discov. 2020;10(7):935–41.
- Varatharaj A, Thomas N, Ellul MA, Davies NWS, Pollak TA, Tenorio EL, et al. CoroNerve Study Group. Neurological and neuropsychiatric complications of COVID-19 in 153 patients: a UK-wide surveillance study. Lancet Psychiatry. 2020;7(10):875–82.
- Hosseini AA, Shetty AK, Sprigg N, Auer DP, Constantinescu CS. Delirium as a presenting feature in COVID-19: Neuroinvasive infection or autoimmune encephalopathy? Brain Behav Immun. 2020;88:68–70.
- Kotfis K, Williams Roberson S, Wilson JE, Dabrowski W, Pun BT, Ely EW. COVID-19: ICU delirium management during SARS-CoV-2 pandemic. Crit Care. 2020;24(1):176.
- Garg RK, Paliwal VK, Gupta A. Encephalopathy in patients with COVID-19: a review. J Med Virol. 2020; https://doi.org/10.1002/ jmv.26207. Epub ahead of print.
- Helms J, Kremer S, Merdji H, Clere-Jehl R, Schenck M, Kummerlen C, et al. Neurologic features in severe SARS-CoV-2 infection. N Engl J Med. 2020;382(23):2268–70.
- Nolan CP, DeAngelis LM. Neurologic complications of chemotherapy and radiation therapy. Continuum (Minneap Minn). 2015;21(2 Neuro-oncology):429–51.
- Neelapu SS, Tummala S, Kebriaei P, Wierda W, Gutierrez C, Locke FL, et al. Chimeric antigen receptor T-cell therapy--assessment and management of toxicities. Nat Rev Clin Oncol. 2018;15(1):47–62.
- Harrison RA, Tummala S, de Groot J. Neurologic toxicities of cancer immunotherapies: a review. Curr Neurol Neurosci Rep. 2020;20(7):27.
- Lee DW, Santomasso BD, Locke FL, Ghobadi A, Turtle CJ, Brudno JN, et al. ASTCT consensus grading for cytokine release syndrome and neurologic toxicity associated with immune effector cells. Biol Blood Marrow Transplant. 2019;25(4):625–38.
- Singer S, Grommes C, Reiner AS, Rosenblum MK, DeAngelis LM. Posterior reversible encephalopathy syndrome in patients with cancer. Oncologist. 2015;20(7):806–11.
- Kamiya-Matsuoka C, Paker AM, Chi L, Youssef A, Tummala S, Loghin ME. Posterior reversible encephalopathy syndrome in cancer patients: a single institution retrospective study. J Neuro-Oncol. 2016;128(1):75–84.
- Major EH, O'Connor P, Mullan B. Single bolus 30% hypertonic saline for refractory intracranial hypertension. Ir J Med Sci. 2015;184(1):159–65.
- 22. Seet RC, Rabinstein AA. Clinical features and outcomes of posterior reversible encephalopathy syndrome following bevacizumab treatment. QJM. 2012;105(1):69–75.
- 23. Miller AH, Monteiro de Oliveira Novaes JA, Brock PA, Sandoval M. Posterior reversible encephalopathy syndrome after bevacizumab treatment presenting to the ED as chest pain and headache. Am J Emerg Med 2016. 1916;34(9):e1–2.
- Ho CH, Lo CP, Tu MC. Bortezomib-induced posterior reversible encephalopathy syndrome: clinical and imaging features. Intern Med. 2014;53(16):1853–7.
- 25. Cai X, Bhattacharyya S, Plitt A, Raibagkar P, LaBuzetta JN, Schleicher SM, et al. Management of posterior reversible encephalopathy syndrome induced by carfilzomib in a patient with multiple myeloma. J Clin Oncol. 2016;34(2):e1–5.

- Magge RS, DeAngelis LM. The double-edged sword: neurotoxicity of chemotherapy. Blood Rev. 2015;29(2):93–100.
- Jain T, Litzow MR. Management of toxicities associated with novel immunotherapy agents in acute lymphoblastic leukemia. Ther Adv Hematol. 2020;11:2040620719899897.
- Strati P, Nastoupil LJ, Westin J, Fayad LE, Ahmed S, Fowler NH, et al. Clinical and radiologic correlates of neurotoxicity after axicabtagene ciloleucel in large B-cell lymphoma. Blood Adv. 2020;4(16):3943–51.
- 29. Locke FL, Ghobadi A, Jacobson CA, Miklos DB, Lekakis LJ, Oluwole OO, et al. Long-term safety and activity of axicabtagene ciloleucel in refractory large B-cell lymphoma (ZUMA-1): a singlearm, multicentre, phase 1-2 trial. Lancet Oncol. 2019;20(1):31–42.
- Nastoupil LJ, Jain MD, Feng L, Spiegel JY, Ghobadi A, Lin Y, et al. Standard-of-care axicabtagene ciloleucel for relapsed or refractory large b-cell lymphoma: results from the US Lymphoma CAR T Consortium. J Clin Oncol. 2020;38(27):3119–28.
- Savica R, Rabinstein AA, Josephs KA. Ifosfamide associated myoclonus-encephalopathy syndrome. J Neurol. 2011;258(9):1729–31.
- 32. Szabatura AH, Cirrone F, Harris C, McDonnell AM, Feng Y, Voit D, et al. An assessment of risk factors associated with ifosfamide-induced encephalopathy in a large academic cancer center. J Oncol Pharm Pract. 2015;21(3):188–93.
- Lo Y, Shen LJ, Chen WH, Dong YH, Wu FL. Risk factors of ifosfamide-related encephalopathy in adult patients with cancer: a retrospective analysis. J Formos Med Assoc. 2016;115(9):744–51.
- Meyers CA, Scheibel RS, Forman AD. Persistent neurotoxicity of systemically administered interferon-alpha. Neurology. 1991;41(5):672–6.
- 35. Caraceni A, Gangeri L, Martini C, Belli F, Brunelli C, Baldini M, et al. Neurotoxicity of interferon-alpha in melanoma therapy: results from a randomized controlled trial. Cancer. 1998;83(3):482–9.
- Somers SS, Reynolds JV, Guillou PJ. Multifocal neurotoxicity during interleukin-2 therapy for malignant melanoma. Clin Oncol (R Coll Radiol). 1992;4(2):135–6.
- Meyers CA, Valentine AD, Wong FC, Leeds NE. Reversible neurotoxicity of interleukin-2 and tumor necrosis factor: correlation of SPECT with neuropsychological testing. J Neuropsychiatry Clin Neurosci. 1994;6(3):285–8.
- van Laar JM, van Buchem MA, Weyl N, Cleton FJ. Reversible neurotoxicity during interleukin-2 therapy for metastatic renal cell carcinoma. Eur J Cancer. 1995;31A(11):1895–7.
- Conry RM, Sullivan JC, Nabors LB 3rd. Ipilimumab-induced encephalopathy with a reversible splenial lesion. Cancer Immunol Res. 2015;3(6):598–601.
- 40. Maur M, Tomasello C, Frassoldati A, Dieci MV, Barbieri E, Conte P. Posterior reversible encephalopathy syndrome during ipilimumab therapy for malignant melanoma. J Clin Oncol. 2012;30(6):e76–8.
- 41. Frantzeskaki F, Rizos M, Papathanassiou M, Nikitas N, Lerikou M, Armaganidis A, et al. L-asparaginase fatal toxic encephalopathy during consolidation treatment in an adult with acute lymphoblastic leukemia. Am J Case Rep. 2013;14:311–4.
- Hourani R, Abboud M, Hourani M, Khalifeh H, Muwakkit S. L-asparaginase-induced posterior reversible encephalopathy syndrome during acute lymphoblastic leukemia treatment in children. Neuropediatrics. 2008;39(1):46–50.
- Fritsch G, Urban C. Transient encephalopathy during the late course of treatment with high-dose methotrexate. Cancer. 1984;53(9):1849–51.
- 44. Lovblad K, Kelkar P, Ozdoba C, Ramelli G, Remonda L, Schroth G. Pure methotrexate encephalopathy presenting with seizures: CT and MRI features. Pediatr Radiol. 1998;28(2):86–91.
- 45. Lee AC, Li CH, Wong YC. Transient encephalopathy following high-dose methotrexate. Med Pediatr Oncol. 2003;41(1):101.

- 46. Packer RJ, Zimmerman RA, Rosenstock J, Rorke LB, Norris DG, Berman PH. Focal encephalopathy following methotrexate therapy. Administration via a misplaced intraventricular catheter. Arch Neurol. 1981;38(7):450–2.
- 47. Ozon A, Topaloglu H, Cila A, Gunay M, Cetin M. Acute ascending myelitis and encephalopathy after intrathecal cytosine arabinoside and methotrexate in an adolescent boy with acute lymphoblastic leukemia. Brain and Development. 1994;16(3):246–8.
- Kuker W, Bader P, Herrlinger U, Heckl S, Nagele T. Transient encephalopathy after intrathekal methotrexate chemotherapy: diffusion-weighted MRI. J Neuro-Oncol. 2005;73(1):47–9.
- 49. Dicuonzo F, Salvati A, Palma M, Lefons V, Lasalandra G, De Leonardis F, et al. Posterior reversible encephalopathy syndrome associated with methotrexate neurotoxicity: conventional magnetic resonance and diffusion-weighted imaging findings. J Child Neurol. 2009;24(8):1013–8.
- Postma TJ, van Groeningen CJ, Witjes RJ, Weerts JG, Kralendonk JH, Heimans JJ. Neurotoxicity of combination chemotherapy with procarbazine, CCNU and vincristine (PCV) for recurrent glioma. J Neuro-Oncol. 1998;38(1):69–75.
- Cumurciuc R, Martinez-Almoyna L, Henry C, Husson H, de Broucker T. Posterior reversible encephalopathy syndrome during sunitinib therapy. Rev Neurol (Paris). 2008;164(6–7):605–7.
- 52. Broniscer A, Leite CC, Lanchote VL, Machado TM, Cristofani LM. Radiation therapy and high-dose tamoxifen in the treatment of patients with diffuse brainstem gliomas: results of a Brazilian cooperative study. Brainstem Glioma Cooperative Group. J Clin Oncol. 2000;18(6):1246–53.
- Maritaz C, Lemare F, Laplanche A, Demirdjian S, Valteau-Couanet D, Dufour C. High-dose thiotepa-related neurotoxicity and the role of tramadol in children. BMC Cancer. 2018;18(1):177.
- 54. Hayase E, Sugita J, Fujimoto K, Ebata K, Yamakawa T, Yoshida M, et al. Posterior reversible encephalopathy syndrome following paralytic ileus caused by vincristine in a patient with T cell lymphoblastic lymphoma. Rinsho Ketsueki. 2014;55(2):249–53.
- 55. Floeter AE, Patel A, Tran M, Chamberlain MC, Hendrie PC, Gopal AK, et al. Posterior reversible encephalopathy syndrome associated with dose-adjusted EPOCH (etoposide, prednisone, vincristine, cyclophosphamide, doxorubicin) chemotherapy. Clin Lymphoma Myeloma Leuk. 2017;17(4):225–30.
- Lawlor PG, Bush SH. Delirium in patients with cancer: assessment, impact, mechanisms and management. Nat Rev Clin Oncol. 2015;12(2):77–92.
- 57. Gabriel Franco J, Santesteban O, Trzepacz P, Bernal C, Valencia C, Ocampo MV, et al. MMSE items that predict incident delirium and hypoactive subtype in older medical inpatients. Psychiatry Res. 2014;220(3):975–81.
- Mayer SA, Coplin WM, Raps EC. Cerebral edema, intracranial pressure, and herniation syndromes. J Stroke Cerebrovasc Dis. 1999;8(3):183–91.
- Byard RW. Patterns of cerebral and cerebellar herniation. Forensic Sci Med Pathol. 2013;9(2):260–4.
- Fisher CM. Brain herniation: a revision of classical concepts. Can J Neurol Sci. 1995;22(2):83–91.
- Wijdicks EF. Neurological picture. Acute brainstem displacement without uncal herniation and posterior cerebral artery injury. J Neurol Neurosurg Psychiatry. 2008;79(7):744.
- 62. Probst MA, Baraff LJ, Hoffman JR, Wolfson AB, Ourian AJ, Mower WR. Can patients with brain herniation on cranial computed tomography have a normal neurologic exam? Acad Emerg Med. 2009;16(2):145–50.

- Sinnott JR, Mohebbi MR, Koboldt T. Papilledema: point-of-care ultrasound diagnosis in the emergency department. Clin Pract Cases Emerg Med. 2018;2(2):125–7.
- Lin AL, Avila EK. Neurologic emergencies in the patients with cancer. J Intensive Care Med. 2017;32(2):99–115.
- Nelson SE, Varelas PN. Status epilepticus, refractory status epilepticus, and super-refractory status epilepticus. Continuum (Minneap Minn). 2018;24(6):1683–707.
- 66. Gutierrez C, Chen M, Feng L, Tummala S. Non-convulsive seizures in the encephalopathic critically ill cancer patient does not necessarily portend a poor prognosis. J Intensive Care. 2019;7:62.
- Hormigo A, Liberato B, Lis E, DeAngelis LM. Nonconvulsive status epilepticus in patients with cancer: imaging abnormalities. Arch Neurol. 2004;61(3):362–5.
- Rojas-Hernandez CM, Oo TH, Garcia-Perdomo HA. Risk of intracranial hemorrhage associated with therapeutic anticoagulation for venous thromboembolism in cancer patients: a systematic review and meta-analysis. J Thromb Thrombolysis. 2017;43(2):233–40.
- 69. Nixon S, Lieberman L, Atenafu EG, Doherty M, Murray C, Wolfe A, et al. Clinical outcomes and transfusion management following intracranial hemorrhage in patients with acute leukemia. Transfusion. 2020;60(2):269–74.
- Khasraw M, Holodny A, Goldlust SA, DeAngelis LM. Intracranial hemorrhage in patients with cancer treated with bevacizumab: the Memorial Sloan-Kettering experience. Ann Oncol. 2012;23(2):458–63.
- Chern JJ, Tsung AJ, Humphries W, Sawaya R, Lang FF. Clinical outcome of leukemia patients with intracranial hemorrhage. J Neurosurg. 2011;115(2):268–72.
- Navi BB, Howard G, Howard VJ, Zhao H, Judd SE, Elkind MSV, et al. New diagnosis of cancer and the risk of subsequent cerebrovascular events. Neurology. 2018;90(23):e2025–e33.
- Zaorsky NG, Zhang Y, Tchelebi LT, Mackley HB, Chinchilli VM, Zacharia BE. Stroke among cancer patients. Nat Commun. 2019;10(1):5172.
- Navi BB, Iadecola C. Ischemic stroke in cancer patients: a review of an underappreciated pathology. Ann Neurol. 2018;83(5):873–83.
- Navi BB, Singer S, Merkler AE, Cheng NT, Stone JB, Kamel H, et al. Recurrent thromboembolic events after ischemic stroke in patients with cancer. Neurology. 2014;83(1):26–33.
- Khasraw M, Posner JB. Neurological complications of systemic cancer. Lancet Neurol. 2010;9(12):1214–27.
- Gitin A, Pfeffer MA, DeMets DL, Hennekens CH. Aspirin in primary prevention: needs individual clinical judgments. Am J Med. 2020;133(7):774–6.
- Murthy SB, Karanth S, Shah S, Shastri A, Rao CP, Bershad EM, et al. Thrombolysis for acute ischemic stroke in patients with cancer: a population study. Stroke. 2013;44(12):3573–6.
- Dardiotis E, Aloizou AM, Markoula S, Siokas V, Tsarouhas K, Tzanakakis G, et al. Cancer-associated stroke: pathophysiology, detection and management (review). Int J Oncol. 2019;54(3):779–96.
- Cutting S, Wettengel M, Conners JJ, Ouyang B, Busl K. Threemonth outcomes are poor in stroke patients with cancer despite acute stroke treatment. J Stroke Cerebrovasc Dis. 2017;26(4):809–15.
- Silvis SM, Hiltunen S, Lindgren E, Jood K, Zuurbier SM, Middeldorp S, et al. Cancer and risk of cerebral venous thrombosis: a case-control study. J Thromb Haemost. 2018;16(1):90–5.
- Petzold A, Smith M. High intracranial pressure, brain herniation and death in cerebral venous thrombosis. Stroke. 2006;37(2):331– 2; author reply 2.
- Ferro JM, Canhão P. Cerebral venous sinus thrombosis: update on diagnosis and management. Curr Cardiol Rep. 2014;16(9):523.



Pituitary Apoplexy

Daria Krivosheya and Ian E. McCutcheon

Case Study

A 78-year-old woman had an intense headache and nausea, both of sudden onset, and consulted her family physician. Noting hypertension to 189/96, he treated her with clonidine and morphine. She presented with unrelieved headache to her local emergency department (ED) the following day, where an MRI showed a sellar and suprasellar mass, the heterogeneity of which suggested hemorrhage within a pituitary macroadenoma (Fig. 17.1). This hemorrhagic component was not, however, recognized at that time. The tumor did not reach the optic chiasm. Cerebral atrophy and chronic ischemic demyelination were also seen, consistent with her age. No endocrine workup was undertaken. Her analgesic was changed to hydromorphone 2 mg by mouth, four times daily. Nonetheless, her headache persisted, and she became forgetful and was excessively sleepy. Her family sought a second opinion 5 days after the ED visit. At that time she was somnolent but had clear mental status when aroused. She said she felt "tipsy" much of the time. She showed no cranial neuropathies, and no visual field deficit. She had tenderness and swelling on the dorsolateral aspect of her right foot but had no recollection of the fall that had caused this local contusion. Her endocrinological history included diabetes mellitus controlled by metformin and Graves' disease with hypothyroidism after 1311 treatment of that condition. She also took methotrexate for rheumatoid arthritis.

On laboratory workup, the following values were obtained: sodium, 119 mEq/L; serum osmolality, 253 mOsm/kg of water; urine osmolality, 217 mOsm/kg of water; magnesium, 1.4 mg/dL; and glucose, 148 mg/dL. Hormonal

D. Krivosheya

I. E. McCutcheon (🖂)



Fig. 17.1 Pituitary macroadenoma with heterogeneity (areas of both hyperintensity and hypointensity) on pre-contrast MRI, sagittal view

assessment showed the following: cortisol, 3.9 mcg/dL (normal, 4.4–12.4); ACTH, <5 pg/mL; prolactin, 11 ng/mL; luteinizing hormone, 1.6 mIU/mL; follicle-stimulating hormone, 12.8 mIU/mL; free T4, 1.20 ng/dL; and thyroidstimulating hormone, 0.49 μ IU/mL. She also had a possible urinary tract infection with 19 WBC/hpf and 7 hyaline casts/ hpf; cultures of urine and blood were negative.

She was admitted to hospital for the following diagnoses: clinically nonfunctional pituitary macroadenoma with pituitary apoplexy; syndrome of inappropriate ADH secretion causing severe hyponatremia; diabetes mellitus, poorly controlled; hypertension; central hypogonadism; cellulitis of the right foot; and headache. Her treatment included transsphenoidal removal of pituitary tumor and associated hemorrhage; fluid restriction; glyburide; adjustment of clonidine; Levaquin x 5 days; discontinuation of hydromorphone; and initiation of hydrocodone/acetaminophen.

Department of Neurological Surgery, Cleveland Clinic Lerner College of Medicine, Akron, OH, USA

Department of Neurosurgery, The University of Texas MD Anderson Cancer Center, Houston, TX, USA e-mail: imccutch@mdanderson.or

K. H. Todd et al. (eds.), Oncologic Emergency Medicine, https://doi.org/10.1007/978-3-030-67123-5_17



Fig. 17.2 Sella (MRI, pre-contrast, sagittal view) showing preservation of pituitary gland after complete removal of pituitary macroadenoma and associated hemorrhage

Transsphenoidal removal of the pituitary tumor was done 10 days after her admission (Fig. 17.2). The tumor was removed in gross total fashion, and in the 6 years since, it has not recurred. Although some cases of pituitary apoplexy demand more immediate surgical intervention, in this patient the tumor was not producing visual loss, and her obtundation was provoked by the unhappy combination of profound hyponatremia and over-medication with a powerful narcotic analgesic. The presence of an unrelated cellulitis required treatment prior to surgery, and time was needed for correction of her hyponatremia, hyperglycemia, and hypocortisolemia in order to reduce surgical risk. Thus, a delayed surgical approach was chosen. Her headaches diminished as serum sodium returned to normal levels and disappeared once the tumor had been removed.

This case offers several important lessons in the management of pituitary apoplexy. It is essential to recognize the MRI signature of hemorrhage in a pituitary adenoma, as the presence of apoplexy raises significantly both the urgency of treating it, and the concomitant risk of pituitary hypofunction. When any pituitary tumor is found, measuring hormone levels and serum electrolytes is essential. Severe headaches caused by a pituitary tumor are best treated by eliminating the tumor, and opioids should not be used at such a high dose in an elderly patient, in whom adverse effects are achieved at lower dose levels than in younger people. It is important to treat any potential inciting causes of pituitary apoplexy (e.g., hypertension or coagulopathy) present when a diagnosis of pituitary apoplexy is made, but treatment is only sufficient when the tumor itself is also treated. Hyponatremia is an important consequence of pituitary apoplexy and should be treated before surgery is undertaken. The same is true for low levels of serum cortisol, which must be corrected with steroid supplementation prior to an operation. Finally, even a complex medical situation in a sick patient with pituitary apoplexy can be managed effectively and safely if a deliberate, thoughtful, careful approach is used, so that surgery can be done effectively and safely.

Definition

Pituitary tumor apoplexy is a clinical syndrome, the hallmarks of which include headache, ophthalmoplegia, and altered level of consciousness. Hemorrhagic infarction of a pituitary tumor leading to rapid expansion of the contents of the sellar region and subsequent acute pituitary gland dysfunction is the underlying basis for this syndrome. The first case of pituitary apoplexy was described by Pierce Bailey in 1898 [1]. However, the term "pituitary apoplexy" was first introduced by Brougham et al. in 1950 in a case report describing five cases of hemorrhagic infarction of pituitary adenomas [2]. The word apoplexy derives from Greek (apoplēssein = to strike down, disable) and means in modern parlance the sudden onset of a neurological deficit, as when a stroke, in a setting of bleeding, results in abrupt loss of function. While nowadays we refer to such bleeding as hemorrhagic stroke, the term apoplexy has remained in use to describe hemorrhage in the pituitary region. Such hemorrhage in most cases occurs when a pituitary tumor is present, and as such, some authors argue that this condition would be more accurately described as pituitary tumor apoplexy [3].

Hemorrhage into a pituitary adenoma with resulting clinical symptoms is also called classical pituitary apoplexy. Spontaneous silent hemorrhage can occur in up to 25% of pituitary adenomas, and when discovered incidentally during routine imaging or histopathologic examinations, it is referred to as subclinical pituitary apoplexy [4, 5]. Pituitary apoplexy has to be distinguished from hemorrhage into a Rathke's cleft cyst, a very rare event whose clinical presentation is identical to pituitary tumor apoplexy syndrome [6–8]. Furthermore, infarction of the pituitary gland during a prolonged or severe period of hypotension is well described in pregnancy and post-partum and is referred to as Sheehan's syndrome [8–10]. The following discussion is dedicated to the diagnosis, management, and outcomes of classical pituitary apoplexy caused by bleeding in a pituitary adenoma.



Fig. 17.3 Post-contrast MRI (coronal, T1-weighted image) showing the anatomical structures in the sella and parasellar region. (1) Optic nerve; (2) suprasellar internal carotid artery; (3) pituitary stalk; (4) lateral wall of cavernous sinus; (5) medial wall of cavernous sinus; (6) sphenoid sinus; (7) pituitary gland; (8) intracavernous carotid artery; (9) abducens nerve (nerve VI); and (10) oculomotor nerve (III nerve)

Anatomy and Physiology Review

To understand the clinical presentation of pituitary apoplexy, it is essential to review the contents of the sellar region and how the apoplectic event disturbs them (Fig. 17.3). Its clinical symptoms can be associated with the sudden onset of one (or more) of four events: presence of intracranial blood; acute onset of pituitary gland dysfunction; loss of vision; and cranial nerve dysfunction, usually manifesting as diplopia.

The presence of acute intracranial blood is a major source of headache. While in most cases of pituitary apoplexy the blood is contained within the tumor, there can be subarachnoid extension of the hemorrhage. Thus, ancillary tests are necessary to rule out a vascular cause of subarachnoid blood such as an aneurysm or a vascular malformation.

The pituitary gland is located within the sella turcica of the sphenoid bone. The gland comprises two parts: an anterior lobe that produces six hormones, including growth hormone (GH), adrenocorticotropic hormone (ACTH), luteinizing hormone (LH), follicle-stimulating hormone (FSH), thyroid-stimulating hormone (TSH), and prolactin, and a posterior lobe, the site of delivery and release of oxytocin and vasopressin produced in the hypothalamus. Acute hemorrhage into the anterior pituitary can result in sudden cessation of secretion of one or more of these hormones. Of the above hormones, acute deficiency of ACTH is of upmost importance as it results in a blunted cortisol response to stress, which if severe enough can be associated with profound hypotension and hyponatremia. Although dysfunction of the posterior pituitary causes alteration of vasopressin secretion leading to central diabetes insipidus and associated hypernatremia, it is rarely observed clinically. Since almost all pituitary adenomas arise in the anterior lobe, they mainly affect the function of that part of the gland. Prolactin, on the other hand, is secreted by the anterior lobe and rises in response to pituitary stalk compression, or when the adenoma itself secretes prolactin. While acute alterations of serum prolactin level do not affect the clinical picture, initial and subsequent prolactin levels have diagnostic significance and thus should be measured.

The two cavernous sinuses are located on either side of the pituitary gland. These venous structures enclose the internal carotid arteries as well as cranial nerves III, IV, and VI. The latter control eye movements, and dysfunction of any of these nerves results in abnormal eye movements on directed gaze with the subjective report of diplopia. Profound compromise of cranial nerve III can also cause eyelid closure. Finally, the optic nerves and chiasm are located in the suprasellar region, and with sufficiently large lesions, compromise of visual acuity and/or fields may occur.

Clinical Presentation

Familiarity with the anatomical structures of the sellar region allows one to understand the symptoms that patients may experience after hemorrhage in that area. The spectrum of presentation of pituitary apoplexy correlates with the size of the hemorrhage and degree of injury to the surrounding structures. While small hemorrhages may be detected radiographically in otherwise asymptomatic patients, a condition referred to as subclinical apoplexy, large hemorrhages can present with an acute alteration of consciousness, cardiovascular collapse, and obtundation, and they can be lifethreatening. Between these two extremes lies a spectrum or constellation of clinical signs that can include headache, changes in vision including diplopia, and various degree of pituitary hypofunction (Fig. 17.4). A given presentation may include one or many of the above signs and symptoms. The most common presenting clinical symptoms are summarized in Table 17.1. The typical ranges of demographics and predisposing or associated factors are shown in Table 17.2 [11].

Headache

Headache is the most common symptom observed in pituitary apoplexy, occurring in 95% of patients [9, 12]. The etiology of headache is likely multifactorial. The rapid increase in pressure within the intrasellar compartment as a result of the hemorrhage causes stretching of the dura lin-



Fig. 17.4 Pituitary macroadenoma with intratumoral hemorrhage (= pituitary apoplexy). This tumor has expanded into the suprasellar space and compresses the optic nerve and chiasm, with resulting onset of severe headache and a temporal visual field deficit. The patient also has profound hypopituitarism. Surgical removal of this hormonally non-functional tumor is indicated

 Table 17.1 Common presenting clinical features of pituitary apoplexy

Clinical symptom
Headache
Nausea and vomiting
Decreased visual acuity
Visual field deficit
Diplopia
Cranial nerve palsy (III, IV, or VI)
Altered mental status
Seizure
Collapse/shock
Coma

ing the sella and produces the resulting headache. Leakage of blood from the hemorrhagic tumor into the subarachnoid space can contribute as well. Vomiting is observed in 70% of patients presenting with a headache and can stem from increased intracranial pressure or from pituitary hypofunction [14]. Headache associated with pituitary apoplexy is most commonly "thunderclap" in nature and tends to respond to resection of the underlying hemorrhagic tumor. When a conservative (non-operative) strategy is chosen, drugs that inhibit platelet aggregation should be avoided and opioid usage limited to prevent clouding of the patient's sensorium [13].

Table 17.2 Clinical correlates of pituitary apoplexy [11]

Gender (male/female) (no.)	48/52
Demographics	(%)
Patients with apoplexy as first presentation of pituitary	73
disease	
Patients with apoplexy and prior known pituitary disease	27
Predisposing/associated factors ^a	
Prior diagnosis of hypertension	33
Pregnant or post-partum	13
Diabetes mellitus	8
Antiplatelet or anticoagulant medication	6
Dopamine agonist	4
Radiotherapy	4
None	46
Secretory status of associated adenoma	
Nonfunctional or gonadotroph adenoma	90
Prolactin-secreting adenoma	10

^aSome patients had more than one predisposing factor

Visual Disturbance

Visual symptoms observed in cases of pituitary apoplexy include either optic nerve dysfunction or diplopia. The optic apparatus, which includes optic nerves, optic chiasm, and optic tracts, is located in the suprasellar compartment and can be subject to direct pressure from the expanding hematoma. Such pressure most commonly provokes bitemporal visual field restriction due to pressure against the undersurface of the chiasm. Direct pressure on one or both nerves or tracts could also result in changes in visual acuity, culminating in a central or junctional scotoma. The latter results in the superior temporal defect in one eye and decreased central vision in the other eye as a result of compression of anterior chiasm, the site of anteriorly looping fibers from the contralateral nasal retina (Fig. 17.5).

Diplopia is another very sensitive presenting symptom of pituitary apoplexy. Cranial nerves that control eye movements, namely III, IV, and VI, run within the cavernous sinus on either side of the sella and are highly susceptible to compression (see Fig. 17.3). Cranial nerve III dysfunction is observed most frequently and can present as ptosis, a dilated pupil and an eye that looks down and out with inability to adduct. Cranial nerve VI innervates the lateral rectus muscle of the eye and controls eye abduction; thus, dysfunction results in inability of the eye to look outward.

Pituitary Dysfunction

Up to 80% of patients with pituitary apoplexy present with pituitary dysfunction [12, 15–17]. While any of the pituitary hormones can be affected, acute decrease in ACTH secretion with subsequent hypocortisolism is the most clinically sig-



Fig. 17.5 Nonfunctional pituitary macroadenoma expanded by hemorrhage and causing a bitemporal visual field deficit. The arrow points to the area of impaction on the optic chiasm. Note the fuzzy, ill-defined hyperintensity present; this is subarachnoid blood that leaked through a disrupted diaphragm sellae. This patient's vision returned to normal after transsphenoidal removal of the tumor and associated hemorrhage

nificant endocrinopathy. Acute adrenal insufficiency can precipitate hypotension (shock) but may also present with hyponatremia, hypoglycemia, nausea, vomiting, obtundation, and coma [18].

Serum prolactin levels at presentation may reflect the degree of pituitary compression and thus give an indirect, albeit imperfect, indication of intrasellar pressure. Low prolactin levels indicate a high degree of pituitary compression and less chance of recovery from hormone impairment after a decompressive procedure [19].

While other pituitary hormones may be affected as a result of pituitary hemorrhage, the reported incidence of these abnormalities varies greatly. Although ACTH dysfunction is most clinically important, it is crucial to identify dysfunction in other hormonal axes as well. GH deficit is seen in 88% of cases, ACTH dysfunction in 66%, TSH abnormality in 42%, and loss of FSH and LH in 85% of cases of pituitary apoplexy [20, 21].

On rare occasions apoplexy causing global infarction of a functional adenoma can cause spontaneous remission of the patient's clinical symptoms of hormonal excess, as well as correction of the relevant hormone levels. This phenomenon has been reported in patients with moderately sized adenomas causing acromegaly or Cushing's disease [22].

Incidence and Predisposing Factors

The true incidence of pituitary apoplexy is difficult to estimate, as many cases are asymptomatic. The reported incidence for symptomatic hemorrhage ranges from 2% to 7% of all pituitary adenomas [12, 15, 16]. In 80% of patients with pituitary apoplexy, the initial diagnosis of hemorrhage is made at the time the tumor is first discovered [23]. There is a slight male predominance for hemorrhage into a pituitary tumor, and most patients present in their fifth or sixth decade [17, 24, 25].

It has been estimated that pituitary adenomas are five times more likely to bleed compared to other tumors [26]. Several characteristics of pituitary adenoma have been identified that put patients at this increased risk of apoplexy. Gender, age, and specific hormonal subtypes carry no increased risk of apoplexy compared to matched controls [21]. An increased risk of hemorrhage is, however, observed in patients with a history of hypertension [12, 17]; usage of anticoagulants or antithrombotic medications [27, 28], estrogen therapy [29], or dopamine agonists [16, 30]; or dynamic testing of pituitary function (see Fig. 17.5) [31, 32]. Rarely, a patient with a pituitary adenoma sustains a head injury that provokes apoplexy [33]. One factor more consistently associated with a higher incidence of apoplexy is tumor size; macroadenomas are much more likely to present with intratumoral hemorrhage than are microadenomas [15, 34–37]. Furthermore, some series have shown that in many cases there is an associated area of infarction within the tumor; thus it is not clear whether the initial event is infarction or hemorrhage [38, 39].

Several hypotheses have been put forward to explain what causes ischemia and hemorrhage in pituitary adenomas. One theory postulates that as a tumor enlarges, it compresses its own blood supply, eventually leading to ischemia and hemorrhage. On the other hand, tumor enlargement may be associated with the tumor's outstripping its own blood supply, thus leading to ischemic necrosis. The higher incidence of hemorrhage into an ischemic or necrotic area within a pituitary adenoma may relate to the unique vascular supply of these tumors [34].

The unique blood supply to the pituitary gland has been implicated in predisposing these tumors to hemorrhage. The pituitary gland is supplied by the hypophyseal portal system. The blood supply to the anterior pituitary comes from the superior hypophyseal arteries, branches of the internal carotid artery that travel along the pituitary stalk and form a rich vascular portal system. The posterior pituitary receives its blood supply from the inferior hypophyseal arteries, which are terminal branches of the meningohypophyseal trunk of the internal carotid artery. There are many anastomoses between the hypophyseal network on and in the pituitary gland and the portal vascular on the pituitary stalk [40]. Pituitary adenomas receive their blood supply from the portal system, as well as directly from the hypophyseal vessels. This exposes the pituitary adenoma to systemic blood pressures. The presence of hypertension with the rich and complex network of the portal system increases the risk of bleeding by fivefold compared to other tumors [13].

Inherent tumor characteristics have also been implicated in predisposing pituitary adenomas to ischemia and apoplexy [41]. Pituitary adenomas are metabolically active tumors that require a continuous supply of glucose and demonstrate a high glucose uptake in 18F-fluorodeoxyglucose-PET studies [42]. Interruption of glucose supply results in rapid adenoma cell death and may lead to infarction [41]. Furthermore, pituitary adenomas have a low level of secretion of such angiogenic factors as vascular endothelial growth factor (VEGF) [43]; thus it is not surprising that they have a reduced vessel density on histological examination which clinically translates into decreased contrast uptake on imaging studies [44]. Finally, perfusion studies have demonstrated very low blood flow in pituitary adenomas [45], likely as a consequence of the high intratumoral pressure that is typical for these tumors [46], factors that predispose pituitary adenomas to a higher risk of developing ischemia. In summary, the combination of the microvascular architecture of pituitary adenomas with the resultant low blood flow and the inherently high metabolic demand of adenoma cells make these tumors very susceptible to ischemic injury during times of systemic blood pressure fluctuation and thus to development of intratumoral pituitary hemorrhage and clinical apoplexy.

Pituitary Apoplexy Workup

Differential Diagnosis

In a patient presenting with a severe headache, a number of other conditions should be considered and ruled out besides pituitary apoplexy. The differential diagnosis includes subarachnoid hemorrhage secondary to a ruptured aneurysm, meningitis, and hydrocephalus, among others. When visual loss and ophthalmoplegia point to pathology in the sellar region, cavernous sinus thrombosis, brain abscess, or a growing cerebral aneurysm should be considered. Other clinical conditions to be considered as part of the differential diagnosis for pituitary apoplexy include temporal arteritis, ophthalmoplegic migraine, hypertensive encephalopathy, basilar artery occlusion, and brainstem stroke or hemorrhage [21, 34].

Diagnostic Imaging

Visualization of the hemorrhage within the sellar region is necessary to make the definitive diagnosis of pituitary apoplexy. Computed tomography (CT) of the brain is a good initial study for detecting acute blood in the sellar region (Fig. 17.6) [47]. A hyperdense lesion in the sellar region on a non-contrast CT scan is highly suggestive of hemorrhage within the first 3 days after the event. Although this modality of imaging is sensitive for hemorrhage, it is not necessarily specific. As intracranial aneurysms and calcifications associated with the craniopharyngioma are also hyperdense on CT scans, it may be difficult to make a definitive diagnosis [35]. To characterize the lesion as well as the underlying pathology, magnetic resonance imaging (MRI) should be obtained [48].

The MRI T2-weighted gradient echo sequence is most sensitive for blood products and may detect even small hemorrhages. It is not, however, accurate at estimating the size or age of the hemorrhage. By surveying the corresponding T1and T2-weighted sequences, it is possible to determine the age of the hemorrhage. The changes in signal intensity of the hemorrhage over time on T1- and T2-weighted MRI scans are summarized in Table 17.3. Furthermore, MRI allows better anatomical visualization of the pituitary gland and may help delineate the underlying tumor, as well as define the relationship of the tumor and the hemorrhage to the optic apparatus. It can also identify extension into either cavernous sinus and rule in or rule out an aneurysm [48–50].

Laboratory Investigations

Fluid and electrolyte disturbances are frequently observed in patients with pituitary apoplexy. Hemorrhage into the pituitary adenoma may affect the function of the pituitary gland itself. In fact, reduced ACTH secretion and resulting hypocortisolemia are observed in 80% of cases [12, 17, 51]. Acute adrenal insufficiency may lead to hyponatremia and, in severe cases, to cardiovascular collapse. Therefore, determination of serum electrolyte levels and fluid balance status is critical to avoid missing adrenal crisis in patients with pituitary apoplexy. Symptoms of acute adrenal insufficiency include vomiting, abdominal pain, myalgia, joint pains, and severe hypotension, leading to hypovolemic shock [18].

Other pituitary hormones can also be affected. TSH deficiency is noted in 50% of cases, and if pre-existing, it may lead to increased morbidity and mortality of pituitary apoplexy in the context of concurrent ACTH deficiency. Disturbance of the gonadotropin axis is observed in 75% of cases [12, 17, 51]. Prolactin levels should be measured, as



Fig. 17.6 Representative images for a patient with pituitary apoplexy. Non-contrast CT axial image (**a**) demonstrates a hyperdense lesion in the sellar region consistent with recent hemorrhage. The T2-weighted axial MRI image (**b**) demonstrates an expansile hyperintense lesion in the sellar region. The T1-weighted coronal slices

through the sellar region (\mathbf{c}, \mathbf{d}) demonstrate a 2.5 cm lesion with an inferior hyperintense component consistent with subacute blood (\mathbf{c}) . The superior component of the tumor heterogeneously enhances with contrast administration, giving an appearance consistent with an underlying pituitary adenoma (\mathbf{d})

 Table 17.3
 Typical appearance of blood on T1- and T2-weighted MRI images

Stage of hemorrhage	Time	T1	T2
Hyperacute	<24 hours	Isointense	Hyperintense
Acute	1-3 days	Isointense	Hypointense
Early subacute	3–7 days	Hyperintense	Hypointense
Late subacute	7-14 days	Hyperintense	Hyperintense
Chronic	>14 days	Hypointense	Hypointense

lower levels of this hormone are predictive of the degree of pituitary compromise and of a higher likelihood of need for pituitary hormone replacement in the future [19].

Alteration of anterior pituitary function is often observed in cases of pituitary apoplexy. In contrast, function of the posterior pituitary gland is rarely affected. While transient hyponatremia in the acute period is frequently attributed to SIADH, permanent central diabetes insipidus is observed in only 3% of cases [52, 53].

In summary, a patient with suspected pituitary apoplexy requires careful assessment of fluid and electrolyte balance. In addition, a complete blood cell count with differential should be obtained to assess for meningitis and for states of coagulopathy secondary to platelet dysfunction. Finally, a panel of hormone studies should be obtained to diagnose and monitor pituitary dysfunction. These should include random serum cortisol, free T4, TSH, IGF-1, and prolactin levels.

Emergency Department Management

Initial management of pituitary apoplexy in the emergency department should focus on supporting the patient's hemodynamic status and treating adrenal insufficiency (Fig. 17.7). The major source of morbidity and mortality associated with pituitary apoplexy is acute adrenal insufficiency from which early series of patients reported mortality close to 50% [2, 34]. However, in recent years there has been quite an improvement in outcomes associated with treatment of this condition largely attributed to improved ability to diagnose pituitary apoplexy with better imaging technology and to recognition and appropriate management of adrenal insufficiency.

The most important initial treatment of pituitary apoplexy and the associated cortisol deficiency is administration of glucocorticoids. After drawing blood for a baseline measurement of serum cortisol, a "stress" dose of 100 mg of hydrocortisone is administered intravenously in patients with suspected adrenal insufficiency. This is followed by a short course of high-dose hydrocortisone (50 mg intravenously at 6-hour intervals) with a subsequent slow taper based on the patient's clinical response. Patients are typically discharged on a maintenance dose of hydrocortisone (15–20 mg daily in divided doses) [18]. In addition to treating the adrenal insufficiency, in the acute setting, glucocorticoids reduce the swelling associated with the underlying tumor and hemorrhage. Thus they minimize the pressure in the sellar region and may lead to symptom improvement.

Intravenous fluid administration to maintain euvolemia and support blood pressure should be instituted. To minimize the amount of edema associated with the hemorrhage, normal saline or Ringer's lactate should be used. If the patient presents with hyponatremia, attention should be paid to the rate of sodium correction to ensure that it increases by 12 mEq/L or less in 24 hours [18]. Gradual correction helps to minimize the risk of central pontine myelinolysis, a devastating complication of sudden large shifts in serum sodium levels.

In patients with pituitary apoplexy who are hypertensive or who take anticoagulants or antithrombotic medications, the blood pressure should be controlled medically and the anticoagulant effects reversed [54]. This is relatively straightforward for patients on heparin or warfarin but more chal-



lenging for those on one of the direct oral anticoagulants now widely used for treating venous thromboembolism and for stroke prevention. These include the direct thrombin inhibitor dabigatran and the Xa inhibitors apixaban, rivaroxaban, and edoxaban. Reversal of the latter group requires either administration of prothrombin complex concentrate or andexanet alfa; dabigatran can be abrogated by idarucizumab (5 gm i.v. over 5 minutes), which binds it with high affinity [55].

Neurosurgical Management

There are no clear guidelines for definitive management of pituitary apoplexy. In a subset of patients managed conservatively, it is difficult to predict who will ultimately require a surgical intervention. No randomized controlled studies are available to provide evidence of differences in outcomes in conservative versus surgically treated patients. Although the question remains unanswered whether operative or nonoperative management is best, patients usually undergo surgery unless their symptoms are minor and no neurological deficits are present (see Fig. 17.7).

Traditionally, pituitary apoplexy has been treated surgically. In patients presenting with acute vision loss, worsening visual field deficit, or ophthalmoplegia, surgical intervention is indicated (see Figs. 17.4 and 17.5) [12, 50, 56]. The "pituitary apoplexy score" (PAS) introduced in 2011 can be used for monitoring patients for signs of deterioration [50]. It incorporates assessment of level of consciousness, visual acuity, visual field deficits, and presence of ocular nerve palsies. A PAS of 4 or greater, or an increasing score while the patient is under observation, may indicate the need for a surgical intervention [50, 57]. Furthermore, surgery should be offered to patients who have no improvement in their symptoms after 1 week of steroid administration [15, 58]. Recent surgical series show that endonasal transsphenoidal surgery for tumor removal and evacuation of hemorrhage provides prompt, effective, and safe decompression of sellar and parasellar structures and is associated with good recovery of vision and clearance of headache [59–61].

Recent retrospective case series demonstrate no difference in the visual and endocrine outcomes between patients treated conservatively and those who underwent surgery and thus make a case for conservative management of pituitary apoplexy [17, 51, 62, 63]. Indeed, some degree of slow shrinkage of tumor has been reported in 76% of patients followed without surgery [59]. These studies, however, suffer from a significant selection bias and do not have appropriately matched controls, since in most studies patients with larger tumors or more severe symptomatology undergo surgical decompression. One study examined the imaging characteristics of pituitary adenoma, including ischemic and hemorrhagic tissue, in an attempt to predict the likelihood of success of conservative management [58]. The presence of a single larger hypodense area within the tumor on a CT scan and early radiographic evidence of tumor involution may be associated with a higher likelihood of clinical resolution of pituitary apoplexy [58]. Regardless of the management strategy chosen, the outcome after treating pituitary apoplexy is excellent, as demonstrated in a recent retrospective cases series of 87 patients [56]. While 20% of patients were treated conservatively, there were no significant differences in outcome metrics between the two groups. Moreover, among those patients, one third presented with a significant alteration of level of consciousness but had an excellent recovery with conservative management [56].

One strong argument for conservative treatment can be made for hemorrhage into a prolactinoma given that medical treatment is the standard of care for that tumor subtype at present. Dopamine agonists are effective at reducing prolactin levels and at reducing the size of the tumor [64]. However, if such a patient presents with worsening vision caused more by the hemorrhage than the tumor volume per se, it is reasonable to consider prompt transsphenoidal decompression of the optic apparatus.

The endoscopic or microsurgical transsphenoidal approach is used most often for decompression of hemorrhage into a pituitary adenoma [34, 61, 65]. This surgery confers low morbidity and mortality for the patient and provides adequate decompression of the neurological structures of the sellar region. Furthermore, because this approach for the most part allows access to the adenoma itself, it allows the surgeon to address the primary lesion in addition to the hemorrhage.

Rarely, subarachnoid bleeding associated with pituitary apoplexy can provoke arterial vasospasm, which may lead to cerebral infarction if not controlled. Vasospasm can be diagnosed by transcranial Doppler or cerebral angiography and reversed by angioplasty directed at the affected vascular segment(s) [66]. As a patient with apoplexy severe enough to cause subarachnoid hemorrhage generally has a concomitant visual and hormonal loss, urgent transsphenoidal decompression is indicated with avoidance of hypotension, in conjunction with angioplasty.

In summary, no clear guidelines exist to determine the optimal management strategy for pituitary apoplexy. A multidisciplinary team involving an endocrinologist, neurosurgeon, and ophthalmologist is necessary to determine the optimal direction of care. Patients with minor symptoms of pituitary apoplexy, or those who have clinical improvement after the apoplectic episode, can be treated conservatively with excellent recovery. On the other hand, patients with visual compromise, significant visual field deficits, and ophthalmoplegia require surgical decompression of the sellar lesion via a transsphenoidal approach.

Outcomes and Follow-Up

The goal of treatment of pituitary apoplexy is to improve compromised visual acuity, reduce visual field defect, and enhance pituitary function. Loss of vision resulting from pressure on the optic nerves has traditionally been thought to be difficult to restore. Nevertheless, if surgical decompression is undertaken within 1 week of the event, there is significant improvement in visual acuity [12, 67]. The success rate is much lower when the surgical intervention is delayed, implying that prolonged compression of optic nerves ultimately results in permanent nerve damage.

In patients who present with minor visual symptoms such as a small field cut, conservative management demonstrated a comparable rate of symptom improvement, thus suggesting that conservative management with glucocorticoids is an alternative strategy [51]. The direct comparison of surgical versus medical management in pituitary apoplexy is complicated due to lack of appropriately matched controls in the studies.

In contrast to the improvement in the visual symptoms, pituitary function does not recover as well after apoplexy. Nevertheless, 50% of patients will have some improvement in pituitary function [19, 68], but 80% of patients will require long-term supplementation of at least one hormone, usually cortisol or thyroxine [12, 51, 62]. Overall, testosterone replacement is needed in 64% of patients [15]. On the other hand, central diabetes insipidus is relatively uncommon, and DDAVP replacement is required only in 20% of patients. Interestingly, the incidence of a need for DDAVP in the conservatively managed group of patients is low, while surgical decompression is associated with a higher rate of diabetes insipidus, observed in 23% of patients treated surgically [56]. Recovery from anterior pituitary insufficiency does seem to correlate somewhat with the prolactin level prior to surgery, with the best outcome in this regard seen when prolactin is ≥ 8.8 ng/mL [69]. That correlation makes sense as a low prolactin indicates a more profoundly damaged anterior lobe.

The pace and likelihood of improvement after resection of an apoplectic pituitary adenoma was nicely demonstrated by Zaidi et al. [70]. In their large surgical series, headache and loss of vision showed consistent resolution, whereas complete correction of ophthalmoplegia occurred in 83%; hormonal deficits recovered infrequently and typically required long-term supplementation. Ophthalmoplegia was the slowest to improve, with a mean time to recovery of 2.4 months, in contrast to the much faster mean times for recovery found with visual field deficits (8 days) and headache (2 days).

Summary

Pituitary apoplexy is a clinical syndrome associated with pituitary tumor hemorrhage that may result in visual acuity and field compromise, ophthalmoplegia, and pituitary dysfunction. Initial investigations should include a CT scan to detect the presence of acute blood, formal visual field testing, and an MRI scan to delineate further the size and extent of the adenoma and hemorrhage. A complete endocrine workup should be performed on admission, and acute cortisol deficiency should be corrected if present. Further management of a patient with pituitary apoplexy should include a multidisciplinary team consisting of a neurosurgeon, endocrinologist, and ophthalmologist, and a decision with respect to surgical versus conservative management should be made in a careful and nuanced fashion driven by the severity of the patient's symptoms, the degree of compression of the optic nerves and chiasm, and the size of the pituitary adenoma. Overall the outcomes of pituitary apoplexy are excellent.

References

- 1. Bailey P. Pathological report of a case of acromegaly, with especial reference to the lesions in the hypophysis cerebri and in the thyroid gland; and a case of hemorrhage into the pituitary. Philadelphia Medical J. 1898;1:789–92.
- Brougham M, Heusner AP, Adams RD. Acute degenerative changes in adenomas of the pituitary body--with special reference to pituitary apoplexy. J Neurosurg. 1950;7(5):421–39.
- Weisberg LA. Pituitary apoplexy. Association of degenerative change in pituitary adenoma with radiotherapy and detection by cerebral computed tomography. Am J Med. 1977;63(1):109–15.
- Bonicki W, Kasperlik-Załuska A, Koszewski W, Zgliczyński W, Wisławski J. Pituitary apoplexy: endocrine, surgical and oncological emergency. Incidence, clinical course and treatment with reference to 799 cases of pituitary adenomas. Acta Neurochir. 1993;120(3–4):118–22.
- Mohr G, Hardy J. Hemorrhage, necrosis, and apoplexy in pituitary adenomas. Surg Neurol. 1982;18(3):181–9.
- Kleinschmidt-DeMasters BK, Lillehei KO, Stears JC. The pathologic, surgical, and MR spectrum of Rathke cleft cysts. Surg Neurol. 1995;44(1):19–26; discussion 26-7.
- Binning MJ, Liu JK, Gannon J, Osborn AG, Couldwell WT. Hemorrhagic and nonhemorrhagic Rathke cleft cysts mimicking pituitary apoplexy. J Neurosurg. 2008;108(1):3–8.
- Schooner L, Wedemeyer MA, Bonney PA, Lin M, Hurth K, Mathew A, et al. Hemorrhagic presentation of Rathke cleft cysts: a surgical case series. Oper Neurosurg (Hagerstown). 2020;18(5):470–9.
- Dash RJ, Gupta V, Suri S. Sheehan's syndrome: clinical profile, pituitary hormone responses and computed sellar tomography. Aust NZ J Med. 1993;23(1):26–31.
- Jemel M, Kandara H, Riahi M, Gharbi R, Nagi S, Kamoun I. Gestational pituitary apoplexy: case series and review of the literature. J Gynecol Obstet Hum Reprod. 2019;48(10):873–81.
- Abbara A, Clarke S, Eng PC, Milburn J, Joshi D, Comninos AN, et al. Clinical and biochemical characteristics of patients presenting with pituitary apoplexy. Endocr Connect. 2018;7(10):1058–66.

- Randeva HS, Schoebel J, Byrne J, Esiri M, Adams CB, Wass JA. Classical pituitary apoplexy: clinical features, management and outcome. Clin Endocrinol. 1999;51(2):181–8.
- Murad-Kejbou S, Eggenberger E. Pituitary apoplexy: evaluation, management, and prognosis. Curr Opin Ophthalmol. 2009;20(6):456–61.
- Suri H, Dougherty C. Presentation and management of headache in pituitary apoplexy. Curr Pain Headache Rep. 2019;23(9):61.
- Bills DC, Meyer FB, Laws ER Jr, Davis DH, Ebersold MJ, Scheithauer BW, et al. A retrospective analysis of pituitary apoplexy. Neurosurgery. 1993;33(4):602–8; discussion 608-9.
- Onesti ST, Wisniewski T, Post KD. Clinical versus subclinical pituitary apoplexy: presentation, surgical management, and outcome in 21 patients. Neurosurgery. 1990;26(6):980–6.
- Sibal L, Ball SG, Connolly V, James RA, Kane P, Kelly WF, et al. Pituitary apoplexy: a review of clinical presentation, management and outcome in 45 cases. Pituitary. 2004;7(3):157–63.
- Charmandari E, Nicolaides NC, Chrousos GP. Adrenal insufficiency. Lancet. 2014;383(9935):2152–67.
- Zayour DH, Selman WR, Arafah BM. Extreme elevation of intrasellar pressure in patients with pituitary tumor apoplexy: relation to pituitary function. J Clin Endocrinol Metab. 2004;89(11):5649–54.
- Veldhuis JD, Hammond JM. Endocrine function after spontaneous infarction of the human pituitary: report, review, and reappraisal. Endocr Rev. 1980;1(1):100–7.
- Ranabir S, Baruah MP. Pituitary apoplexy. Indian J Endocrinol Metab. 2011;15(Suppl3):S188–96.
- Roerink SH, van Lindert EJ, van de Ven AC. Spontaneous remission of acromegaly and Cushing's disease following pituitary apoplexy: two case reports. Neth J Med. 2015;73(5):242–6.
- Biousse V, Newman NJ, Oyesiku NM. Precipitating factors in pituitary apoplexy. J Neurol Neurosurg Psychiatry. 2001;71(4):542–5.
- 24. Reid RL, Quigley ME, Yen SS. Pituitary apoplexy. A review. Arch Neurol. 1985;42(7):712–9.
- McFadzean RM, Doyle D, Rampling R, Teasdale E, Teasdale G. Pituitary apoplexy and its effect on vision. Neurosurgery. 1991;29(5):669–75.
- Wakai S, Fukushima T, Teramoto A, Sano K. Pituitary apoplexy: its incidence and clinical significance. J Neurosurg. 1981;55(2):187–93.
- Willamowicz AS, Houlden RL. Pituitary apoplexy after anticoagulation for unstable angina. Endocr Pract. 1999;5(5):273–6.
- Nagarajan DV, Bird D, Papouchado M. Pituitary apoplexy following anticoagulation for acute coronary syndrome. Heart. 2003;89(1):10.
- Cardoso ER, Peterson EW. Pituitary apoplexy: a review. Neurosurgery. 1984;14(3):363–73.
- Knoepfelmacher M, Gomes MC, Melo ME, Mendonca BB. Pituitary apoplexy during therapy with cabergoline in an adolescent male with prolactin-secreting macroadenoma. Pituitary. 2004;7(2):83–7.
- Levy A. Hazards of dynamic testing of pituitary function. Clin Endocrinol. 2003;58(5):543–4.
- Lee DH, Chung MY, Chung DJ, Kim JM, Lee TH, Nam JH, Park CS. Apoplexy of pituitary macroadenoma after combined test of anterior pituitary function. Endocr J. 2000;47(3):329–33.
- Ward M, Kamal N, Majmundar N, Baisre-De Leon A, Eloy JA, Liu JK. Post-traumatic pituitary tumor apoplexy after closed head injury: case report and review of the literature. World Neurosurg. 2018;120:331–5.
- Nawar RN, AbdelMannan D, Selman WR, Arafah BM. Pituitary tumor apoplexy: a review. J Intensive Care Med. 2008;23(2):75–90.
- 35. Verrees M, Arafah BM, Selman WR. Pituitary tumor apoplexy: characteristics, treatment, and outcomes. Neurosurg Focus. 2004;16(4):E6.

- Mohanty S, Tandon PN, Banerji AK, Prakash B. Haemorrhage into pituitary adenomas. J Neurol Neurosurg Psychiatry. 1977;40(10):987–91.
- Rolih CA, Ober KP. Pituitary apoplexy. Endocrinol Metab Clin N Am. 1993;22(2):291–302.
- Semple PL, De Villiers JC, Bowen RM, Lopes MB, Laws ER Jr. Pituitary apoplexy: do histological features influence the clinical presentation and outcome? J Neurosurg. 2006;104(6):931–7.
- Lubina A, Olchovsky D, Berezin M, Ram Z, Hadani M, Shimon I. Management of pituitary apoplexy: clinical experience with 40 patients. Acta Neurochir. 2005;147(2):151–7; discussion 157.
- Flerkó B. Fourth Geoffrey Harris memorial lecture: the hypophysial portal circulation today. Neuroendocrinology. 1980;30(1):56–63.
- Oldfield EH, Merrill MJ. Apoplexy of pituitary adenomas: the perfect storm. J Neurosurg. 2015;122(6):1444–9.
- 42. Francavilla TL, Miletich RS, DeMichele D, Patronas NJ, Oldfield EH, Weintraub BD, Di Chiro G. Positron emission tomography of pituitary macroadenomas: hormone production and effects of therapies. Neurosurgery. 1991;28(6):826–33.
- 43. Berkman RA, Merrill MJ, Reinhold WC, Monacci WT, Saxena A, Clark WC, et al. Expression of the vascular permeability factor/ vascular endothelial growth factor gene in central nervous system neoplasms. J Clin Invest. 1993;91(1):153–9.
- 44. Schechter J. Ultrastructural changes in the capillary bed of human pituitary tumors. Am J Pathol. 1972;67(1):109–26.
- Kruse A, Astrup J, Cold GE, Hansen HH. Pressure and blood flow in pituitary adenomas measured during transsphenoidal surgery. Br J Neurosurg. 1992;6(4):333–41.
- 46. Arafah BM, Prunty D, Ybarra J, Hlavin ML, Selman WR. The dominant role of increased intrasellar pressure in the pathogenesis of hypopituitarism, hyperprolactinemia, and headaches in patients with pituitary adenomas. J Clin Endocrinol Metab. 2000;85(5):1789–93.
- 47. Post MJ, David NJ, Glaser JS, Safran A. Pituitary apoplexy: diagnosis by computed tomography. Radiology. 1980;134(3):665–70.
- 48. Goyal P, Utz M, Gupta N, Kumar Y, Mangla M, Gupta S, Mangla R. Clinical and imaging features of pituitary apoplexy and role of imaging in differentiation of clinical mimics. Quant Imaging Med Surg. 2018;8(2):219–31.
- Lazaro CM, Guo WY, Sami M, Hindmarsh T, Ericson K, Hulting AL, Wersäll J. Haemorrhagic pituitary tumours. Neuroradiology. 1994;36(2):111–4.
- Rajasekaran S, Vanderpump M, Baldeweg S, Drake W, Reddy N, Lanyon M, et al. UK guidelines for the management of pituitary apoplexy. Clin Endocrinol. 2011;74(1):9–20.
- Ayuk J, McGregor EJ, Mitchell RD, Gittoes NJ. Acute management of pituitary apoplexy--surgery or conservative management? Clin Endocrinol. 2004;61(6):747–52.
- 52. Möller-Goede DL, Brändle M, Landau K, Bernays RL, Schmid C. Pituitary apoplexy: re-evaluation of risk factors for bleeding into pituitary adenomas and impact on outcome. Eur J Endocrinol. 2011;164(1):37–43.
- Sweeney AT, Blake MA, Adelman LS, Habeebulla S, Nachtigall LB, Duff JM, Tully GL 3rd. Pituitary apoplexy precipitating diabetes insipidus. Endocr Pract. 2004;10(2):135–8.
- Santos AR, Bello CT, Sousa A, Duarte JS, Campos L. Pituitary apoplexy following systemic anticoagulation. Eur J Case Rep Intern Med. 2019;6(12):001254.
- Langer A, Connors JM. Assessing and reversing the effect of direct oral anticoagulants on coagulation. Anesthesiology. 2020;133(1):223–32.
- Singh TD, Valizadeh N, Meyer FB, Atkinson JL, Erickson D, Rabinstein AA. Management and outcomes of pituitary apoplexy. J Neurosurg. 2015;122(6):1450–7.

- 57. Bujawansa S, Thondam SK, Steele C, Cuthbertson DJ, Gilkes CE, Noonan C, et al. Presentation, management and outcomes in acute pituitary apoplexy: a large single-centre experience from the United Kingdom. Clin Endocrinol. 2014;80(3):419–24.
- Maccagnan P, Macedo CL, Kayath MJ, Nogueira RG, Abucham J. Conservative management of pituitary apoplexy: a prospective study. J Clin Endocrinol Metab. 1995;80(7):2190–7.
- 59. Gondim JA, de Albuquerque LAF, Almeida JP, Bulcao T, Gomes E, Schops M, et al. Endoscopic endonasal surgery for treatment of pituitary apoplexy: 16 years of experience in a specialized pituitary center. World Neurosurg. 2017;108:137–42.
- 60. Almeida JP, Sanchez MM, Karekezi C, Warsi N, Fernández-Gajardo R, Panwar J, et al. Pituitary apoplexy: results of surgical and conservative management clinical series and review of the literature. World Neurosurg. 2019;130:e988–99.
- Pangal DJ, Chesney K, Memel Z, Bonney PA, Strickland BA, Carmichael J, et al. Pituitary apoplexy case series: outcomes after endoscopic endonasal transsphenoidal surgery at a single tertiary center. World Neurosurg. 2020;137:e366–72.
- Gruber A, Clayton J, Kumar S, Robertson I, Howlett TA, Mansell P. Pituitary apoplexy: retrospective review of 30 patients--is surgical intervention always necessary? Br J Neurosurg. 2006;20(6):379–85.
- 63. Sahyouni R, Goshtasbi K, Choi E, Mahboubi H, Le R, Khahera AS, et al. Vision outcomes in early versus late surgical interven-

tion of pituitary apoplexy: meta-analysis. World Neurosurg. 2019;127:52–7.

- Arafah BM, Nasrallah MP. Pituitary tumors: pathophysiology, clinical manifestations and management. Endocr Relat Cancer. 2001;8(4):287–305.
- Johnston PC, Hamrahian AH, Weil RJ, Kennedy L. Pituitary tumor apoplexy. J Clin Neurosci. 2015;22(6):939–44.
- 66. Douleh DG, Morone PJ, Mobley B, Fusco MR, Chambless LB. Angioplasty is an effective treatment for vasospasm following pituitary apoplexy and tumor resection. Cureus. 2018;10(1):e2117.
- Agrawal D, Mahapatra AK. Visual outcome of blind eyes in pituitary apoplexy after transphenoidal surgery: a series of 14 eyes. Surg Neurol. 2005;63(1):42–6; discussion 46.
- Arafah BM, Harrington JF, Madhoun ZT, Selman WR. Improvement of pituitary function after surgical decompression for pituitary tumor apoplexy. J Clin Endocrinol Metab. 1990;71(2):323–8.
- 69. Lammert A, Walter MS, Giordano FA, Al Zhgloul M, Krämer BK, Nittka S, et al. Neuro-endocrine recovery after pituitary apoplexy: prolactin as a predictive factor. Exp Clin Endocrinol Diabetes. 2020;128(5):283–9.
- Zaidi HA, Cote DJ, Burke WT, Castlen JP, Bi WL, Laws ER Jr, Dunn IF. Time course of symptomatic recovery after endoscopic transsphenoidal surgery for pituitary adenoma apoplexy in the modern era. World Neurosurg. 2016;96:434–9.

Spinal Cord Compression

Jayne M. Viets-Upchurch and Sorayah S. Bourenane

Case Study

A 73-year-old woman with a history of breast cancer presented to the emergency department (ED) for evaluation of bilateral lower extremity weakness and shaking. She was diagnosed with locally invasive, nonmetastatic breast cancer 6 months prior to her presentation. Treatment has been delayed due to endemic health concerns. In the interval, she received hormonal therapy in anticipation of surgical resection which was scheduled to take place in 2 days.

She reported a 3-week history of bilateral lower extremity weakness with a sensation that her knees were about to give out. Symptoms were intermittently and without warning. Although she has not fallen, her symptoms occur frequently enough that she quit taking her routine morning walks. Two weeks prior to her presentation, she developed bilateral hip pain that was sharp and shooting. It occurred at night and was described as nonradiating and 8/10 in severity. Three days prior to her presentation, she had a near-miss fall in her kitchen and now feels weak enough that she ambulates with a cane. For the past 2 days, her calf muscles felt tight bilaterally. She denies fecal or urinary incontinence and has no numbness or paresthesias. She has a headache she describes as "not bad." She has taken no medications for her symptoms, electing to treat them with breathing exercises, meditation, and rest.

She was evaluated at an urgent care facility in her hometown. The physician suggested she might have Guillain-Barre syndrome and offered no further evaluation or therapy.

J. M. Viets-Upchurch (🖂)

Department of Emergency Medicine, Division of Internal Medicine, The University of Texas MD Anderson Cancer Center, Houston, TX, USA e-mail: jviets@mdanderson.org

S. S. Bourenane

She traveled 1500 miles yesterday to arrive at your facility for her surgery day after tomorrow.

Aside from lower extremity ankle edema after air travel yesterday, her review of systems is otherwise negative. Her past history is notable for venous thromboembolism managed with IVC filter placement.

Her physical exam reveals a well-appearing woman. Her neurologic exam revealed normal strength and patellar reflexes with intact sensation, with no saddle anesthesia or clonus. Her gait is not tested due to the patient's sensation of instability. The remainder of her examination is unremarkable, with the exception of a firm tumor mass in the upper outer quadrant of the left breast. There is no apparent lymphadenopathy and only mild ankle edema.

Her laboratory evaluation reveals a normal CBC, metabolic profile, and no evidence of coagulopathy. A CT of the head is normal; however, an MRI of the cervical thoracic and lumbar spines shows evidence of metastases at T9 with circumferential epidural tumor burden compressing and deforming the cord. She has further osseous metastases at L5 with circumferential epidural tumor burden narrowing the canal and crowding the roots of the cauda equina.

Neurosurgery evaluated the patient in the ED. She had no bony impingement upon the spinal cord or thecal sac, and recommendations were given for radiation therapy. Radiation oncologic evaluated the patient and deemed her a suitable candidate for therapy.

The patient was admitted to the hospital and treated with steroids and analgesics. On restaging, a CT of the chest showed the known primary tumor with new invasion of the chest wall musculature as well as scattered pulmonary nodules too small to characterize. There was no lymphadenopathy. CT of the abdomen and pelvis revealed no clear-cut evidence of visceral metastases. Multiple bony lesions were noted. Nuclear medicine bone scan was notable for multifocal skeletal uptake consistent with metastases at T9, L5, left acetabulum, left femur, and left ischium.

Physical therapy was consulted. She was discharged home on hospital day 3 with improved symptoms. Outpatient



Emergency Department & Clinical Decision Unit, The University of Texas MD Anderson Cancer Center, Houston, TX, USA

K. H. Todd et al. (eds.), Oncologic Emergency Medicine, https://doi.org/10.1007/978-3-030-67123-5_18

follow-up was scheduled with physical therapy, relation oncologic, and breast medical oncologic.

Introduction

Traumatic spinal cord injury is a well-documented, devastating injury leading to sensory loss, paralysis, sphincter dysfunction, and protracted pain. The Centers for Disease Control and Prevention estimates that there are 12,000– 20,000 such cases annually [1]. Malignant spinal cord compression (MSCC) is equally devastating and occurs at a similar rate to traumatic spinal cord injury [2]. Curative and life-prolonging treatments in cancer patients are expected to yield an increased incidence of the condition.

Cancer is the most common systemic disease affecting the spine [3]. MSCC presents as a true oncologic emergency resulting from tumor-related compression of the thecal sac and spinal cord [2, 4]. As with traumatic injury, untreated MSCC leads to paraplegia, incontinence, and permanent disability. It presents clinically in approximately 3–10% of cancer patients prior to death [5, 6]. Cancer patients have a median survival of 3–6 months from diagnosis of MSCC [5, 7, 8]. Once treated, 8–20% of patients experience recurrent compression either locally or at another vertebral level [9, 10].

The past several years have yielded significant advances in the diagnosis and treatment of this condition. Prompt recognition and treatment is essential to maintain mobility and neurological function. Any former or active cancer patient presenting with new or worsening back pain, even in the absence of neurological deficit, merits evaluation for spinal cord compression [11]. Individual risk assessment is necessary [12].

Epidemiology

Postmortem studies suggest that MSCC is present in up to 36% of cancer patients [13, 14]. In a US nationwide study of 15,367 cases of MSCC, the mean age at hospitalization was 62 years [6]. Men outnumber women nearly two to one, with only 37.5% of cases occurring in women. In approximately 20% of cases, MSCC is the initial presentation of cancer [8, 15]. Increasing cancer prevalence, coupled with prolonged life expectancy of diagnosed patients, is anticipated to drive the incidence upward.

Prevalence of MSCC varies depending on tumor type. Cancers of the breast, lung, and prostate each account for 15–20% of cases. Renal cell carcinoma accounts for an additional 5–10%. Multiple myeloma and lymphoma are the most frequent liquid tumors associated with MSCC, each contributing 5–10% of cases [2, 6, 16]. MSCC affects up to 15% of myelomas [5, 6] and 13% of lymphomas [6]. In patients with prostate cancer, 5.5% develop MSCC [6]. In contrast, it occurs in only 0.2% of pancreatic cancers [5]. In children, the most commonly associated malignancies are sarcoma, neuroblastoma, germ-cell tumor, and Hodgkin lymphoma [2, 16].

The most common location of MSCC is the thoracic spine (70% of cases); 20% of cases occur at the lumbosacral level and 10% in the cervical region [2]. It is hypothesized that this pattern represents the primary tumor's lymphatic drainage. Accordingly, metastases from breast and lung cancers tend to be found in the thoracic spine. Pelvic and intra-abdominal malignancies most commonly migrate to the lumbar spine. Multiple, noncontiguous spinal epidural metastases were noted in 32% of patients who underwent complete imaging of the spine [17]. Among those, two-thirds have multicentric disease affecting more than one region (cervical, thoracic, and lumbar) of the spine [18].

Historical factors that predispose the patient to MSCC include known metastatic disease at the time of cancer diagnosis and known vertebral metastases.

Pathophysiology

Most cases of MSCC are epidural in origin, arising from the vertebral column in 85% of patients [2]. The marrow of the vertebral body provides fertile ground for circulating tumor cells. Their growth eventually replaces the marrow and invades into the canal through thrombosed venous plexuses. These events appear to occur independently of bony destruction [19]. Epidural spread may also occur by direct tumor extension or by direct deposition of tumor cells into the epidural space [16].

Ultimately, neuronal injury is thought to involve vasogenic edema [20] leading to ischemia [19], presumably through venous infarction, but there has been debate regarding this last phenomenon [21]. In cases of paralysis, demyelination is evident [21].

Clinical Presentation

History

Presenting symptoms of MSCC can be motor (weakness, paraplegia), sensory (pain, numbness, neuropathy), and/or autonomic (incontinence). Symptoms vary based on the location of the metastatic lesion; however, they are a poor indicator of the level of involvement [18].

Although it is nonspecific, pain is the primary complaint in 83–96% of those with MSCC [22, 23]. Pain can be referred, local, radicular, or a combination of all three [24]. Misdiagnosis is a common issue in the ED setting. In an interesting retrospective study of 63 patients with spinal cord compression from a variety of causes (only 10 were due to cancer), 18 (29%) were misdiagnosed [25]. Consequently, there was a significant delay in diagnosis despite obvious neurological deficits at presentation. Highlighting this historical delay, back pain is present for a median of 62 days prior to treatment of MSCC [26].

Patients may report pain in a band-like distribution. It is generally described as sharp, shooting, or deep [11]. As with mechanical causes of back pain, the pain associated with MSCC may worsen with weight-bearing loads, which bring pressure to bear on the vertebral column [27]. Other common precipitating factors include coughing, bending, and sneezing [11]. Twenty percent of patients report that rest in a supine position exacerbates symptoms and often disrupts sleep [3, 11]. These patients frequently sleep in an upright position.

Weakness follows pain with an estimated 35–85% of patients endorsing the symptom [28]. Patients presenting with radiculopathy are symptomatic an average of 9 weeks prior to diagnosis [29].

Early studies showed ataxia in up to 67% of patients with nearly 65% of those having profound motor deficits that prevented walking [22]. Perhaps as a result of heightened awareness, a more recent study of patients receiving radiation therapy showed 67% of patients were ambulatory (although 28% required assistance) [30]. Loss of sensation, dense paraplegia, and incontinence are late findings and signal some degree of permanent disability [22].

Because of its high specificity (98%), any cancer patient with new back pain should be considered to have metastasis until proven otherwise [3]. Because cord compression is an evolving condition, a patient with previously stable back pain may present with recently worsening symptoms.

Physical Examination

Spinal tenderness may be present overlying the level of metastatic deposit. Nonetheless, it is difficult to thoroughly assess the location of MSCC lesion on the basis of examination alone. Pain in the thoracic spine and abnormal gait are suggestive of cord compression [31]. However, the absence of these findings is not sufficient to rule out the disease.

Documentation of a detailed physical examination is essential. Serial exams are frequently required, and a thorough neurologic exam, including sensation, strength, and reflexes, should be carefully recorded upon presentation.

Patients with cervical spine tenderness or symptomatology should be immobilized and placed in a Philadelphia collar until stability of the area can be assessed. Likewise, if

 Table 18.1
 Levels of disability according to Frankel et al. [32]

Frankel grade	Motor	Sensory
А	Complete paresis	Insensate
В	Complete paresis	Sensation preserved
С	Incomplete but nonfunctional	Sensation preserved
D	Functional but symptomatic	Sensation preserved
E	Normal	Normal

spinal instability is suspected, range-of-motion testing is contraindicated, and the patient should be immobilized.

Hyperreflexia and an upward-going Babinski reflex are common findings [31]. Weakness and paraplegia are late findings. Decreased rectal sphincter tone and urinary incontinence are indicators of poor outcome. A bedside bladder scan may be useful for documenting post-void residual to assess for evidence of urinary retention.

Degree of disability is typically described using the Frankel grading scale [32]. This scale, established in 1969, classifies patients according to five levels (A through E) with A being complete loss of both motor and sensory function and E being normal (Table 18.1).

Imaging

Advanced imaging is essential to delineate the extent of disease. By themselves, plain films of the spine are of little value in diagnosing the condition. Approximately 26–29% of metastatic deposits are occult on plain film X-ray [8, 14]. Furthermore, given the prevalence of osteoarthritic changes of the spine in adult patients, such plain films may offer false reassurance as to the etiology of the pain.

Prior to the 1990s, spinal cord compression was diagnosed by myelography [33]. Fortunately, this invasive and uncomfortable procedure has largely been supplanted by magnetic resonance imaging (MRI) (Fig. 18.1) [18].

While gadolinium-enhanced MRI can help to determine intradural tumor or leptomeningeal disease, it is not required for cord compression studies. Unenhanced MRI is equal to myelography in detecting epidural disease and is more sensitive at detecting vertebral metastasis [34], justifying its use and reducing procedure time compared to gadoliniumenhanced studies.

Computed tomography (CT)-guided myelography is of value for patients unable to tolerate MRI [35]. If both CT and MRI are unavailable, bone scintigraphy in combination with plain films may be of use [33].

Imaging studies should include the entire spine, not just the perceived area of pain. Up to 31% of patients have multilevel disease [17]. Sensory deficits and mechanical pain may be present 2–4 vertebral levels away from the actual lesion [31]. If MRI suggests cord compression, severity can be graded using the Epidural Spinal Cord Compression scale described by Bilsky [36]. This 6-point scale rates the degree of cord compression between 1 and 3 (with 3 being the most severe) based on the level of cord impingement (Fig. 18.2) [37]. A recent trial confirmed its continued utility for enhancing agreement and clarity of communication among clinicians [38].



Fig. 18.1 A 16-year-old female with ovarian cancer and spinal cord compression at the level of the third thoracic vertebra

Management

The goal of therapy is symptom control and preservation of function. This requires a multidisciplinary approach involving nursing, medical oncologic, neurosurgery, and radiation oncologic services.

Nursing Efforts

Upon diagnosis and initiation of therapy, serial neurological evaluations should be undertaken. Neuro-vital signs should be scheduled to coincide with other nursing efforts to minimize patient discomfort and ease the burden of care. Patients with involvement of the cervical spine should have a Philadelphia collar placed until spinal stability has been confirmed. Strict bed rest (including logroll and bedpan use) should be instituted if there is suspicion of spinal cord instability.

In the United Kingdom, the National Institutes for Health Care Excellence guidelines recommend that all patients with suspected cord compression be maintained in a flat position [29]. Other authors do not believe that strict bed rest is necessary. Proponents of this theory believe that MSCC is inherently different from that caused by trauma. Authors supporting this position contend that the increased incidence of deep vein thrombosis, infection (particularly from the respiratory and urinary tracts), and decubitus ulcers outweigh the benefit of bed rest.

There is a lack of data or randomized controlled trials to strongly support either position [39]. A recent systematic review of non-pharmacological interventions in spinal cord



Fig. 18.2 Schematic representation of the 6-point ESCC grading scale. A grade of 0 indicates bone-only disease; 1a, epidural impingement, without deformation of the thecal sac; 1b, deformation of the thecal sac, without spinal cord abutment; 1c, deformation of the thecal

sac with spinal cord abutment but without cord compression; 2, spinal cord compression but with CSF visible around the cord; and 3, spinal cord compression, no CSF visible around the cord. (From Bilsky et al. [37], with permission Elsevier)
compression found that while supine positioning is extremely important in other causes of spinal cord compression, it seemed excessive in MSCC patients, although again, the quality of the data was poor [40]. Patient performance status and preference should be taken into consideration as those with good functional status may be quite resistant to bed rest, particularly if it worsens their symptoms.

Medical Therapy

The mainstay of medical therapy is treatment with corticosteroids [4]. Unless contraindicated, they are recommended for all patients, particularly those with neurological deficits. Initial trials demonstrated that corticosteroids improve functional status, as well as symptoms, in MSCC.

Controversy exists regarding the effective dose. A randomized, controlled trial by Sorensen et al. evaluated functional outcomes in 57 patients who received radiotherapy in conjunction with high-dose versus no corticosteroids as adjunctive therapy. Fifty-nine percent of patients in the dexamethasone group were ambulatory 6 months after treatment compared to 39% in the group who did not receive steroids [41].

More recent studies have shown similar results with lower-dose corticosteroids [42]. A recent systematic review by Kumar suggests a loading dose of 10 mg of intravenous dexamethasone followed by 16 mg per day orally in two to four divided doses [4, 5, 43, 44]. In patients with recentonset neurological deficits, higher doses may be considered [4]. National guidelines in both Canada and England now recommend more moderate dosing.

Corticosteroids are associated with psychosis, gastric ulceration/perforation, rectal bleeding, and hyperglycemia. Care should be taken to mitigate against these effects. Steroids should be weaned as early as tolerated and gastrointestinal prophylaxis should be initiated. Special attention should be given to glucose control (particularly in patients with preexisting diabetes). While not typically of use in the ED, chemotherapy is of value in longer-term management. In particular, seminomas, myeloma, and lymphomas may show a dramatic response to treatment [45–47].

Surgical Therapy

Overall, decompressive surgery is indicated in 10–15% of MSCC cases [48]. Surgical evaluation is required in cases of spinal instability, direct cord compression due to a bony fragment, and impending sphincter dysfunction. Unknown primary tumors will require biopsy even if full resection is not possible. Patients not responding to radiotherapy or those who have previously received radiotherapy may benefit from surgical intervention [20]. Radoresistant tuores, including

sarcoma, melanoma, GI tumors, lung, renal and thyroid, are more likely to require surgical intervention [49].

The Spinal Instability Neoplastic Score (SINS) developed by Fisher et al. is the most frequently used tool to determine patients that will require surgical stabilization. The scale assigns a score between 0 and 18 to metastatic deposits. Six elements are considered, including lesion location, the degree of pain, type of lesion, radiographic spinal alignment, degree of vertebral body collapse, and the presence of involvement of the posterolateral spinal elements. Scores of less than 6 are generally considered stable. Those greater than 13 are deemed unstable and likely to require surgical intervention. Indeterminate lesions graded 7–12 require further evaluation [50] (Table 18.2).

Patients with good functional status, limited disease, and a life expectancy greater than 3–6 months may benefit from surgery [51]. Those with paraplegia of less than 48 hours duration may experience a degree of functional restoration. In a recent study, 101 patients presenting with neurologic symptoms underwent decompressive laminectomy. 74% showed improved motor function and 51% had regained the ability to walk at discharge [52].

Previously, laminectomy was the primary surgical option available. In many cases, it provided suboptimal results. Over

 Table 18.2
 Spinal Instability Neoplastic Score (SINS) System (From Fisher et al. [50], with permission Wolters Kluwer Health)

Component	Score
Location	
Junctional (occiput-C2, C7-T2, T11-L1, L5-S1)	3
Mobile spine (C3–6, L2–4)	2
Semi-rigid (T3-10)	1
Rigid (S2–S5)	0
Pain relief with recumbency and/or pain with movement/	
loading of the spine	
Yes	3
No (occasional pain but not mechanical)	1
Pain-free lesion	0
Bone lesion	
Lytic	2
Mixed (lytic/blastic)	1
Blastic	0
Radiographic spinal alignment	
Subluxation/translation present	4
De novo deformity (kyphosis/scoliosis)	2
Normal alignment	0
Vertebral body collapse	
>50% collapse	3
<50% collapse	2
No collapse with >50% body involved	1
None of the above	0
Posterolateral involvement of the spinal elements (facet, pedicle, or CV joint fracture or replacement with tumor)	
Bilateral	3
Unilateral	1
None of the above	0

the past two decades, significant improvements have been made. New surgical techniques including circumferential decompression of the spine with anterior posterior surgery stabilization have advanced the care of these patients [53].

Minimally invasive procedures, in conjunction with stereotactic body radiation therapy, have shown promise in the management of MSCC [54]. Although not routinely used, vertebroplasty or kyphoplasty has been shown to be effective for management of pain associated with pathologic compression fractures. Some authors have recommended kyphoplasty with stabilization. Chen reported excellent results in a study of 282 patients undergoing percutaneous kyphoplasty for metastatic disease. Patients were followed as far out as 15 months and reported decreased use of analgesia as well as improved quality of life and Karnofsky performance scores [55]. Similarly, in a study of 50 patients with spinal metastasis, Kwan found minimally invasive stabilization with percutaneous screws provided adequate relief of symptoms, with an associated improvement in Frankel score [56]. In a study specific to patients with MSCC, Chen found kyphoplasty with short instrumentation provided equivalent results to more traditional long stabilization. Moreover, there was less blood loss and fewer complications [57].

Prior to any surgical procedure, any coagulopathy should be corrected, and thrombocytopenic patients should receive platelet infusion. A recent review found post-operative complications are common. These include wound infection/ dehiscence, pneumonia, pleural effusion, respiratory failure, pulmonary embolism, venous thrombosis, cerebral spinal fluid leak, urinary tract infection, and instrument failure [58].

Radiotherapy

The combination of surgery with radiotherapy has been shown to improve outcomes [59]. Whether for therapeutic or palliative intent, radiotherapy is provided to virtually all patients.

Tumors such as leukemia, lymphoma, and germ-cell tumors are particularly responsive to radiotherapy. Breast, prostate, and ovarian cancers have intermediate sensitivity [49].

Issues regarding duration and dosing depend on intent of therapy and must be individualized to the patient. While guidelines have been recommended, there are few randomized, controlled trials to clarify specific regimens [60].

Generally speaking, patients with a favorable prognosis will benefit from longer courses of therapy [61, 62]. End-of-life patients with a poor prognosis are typically treated with a single-dose regimen [61, 63]. These regimens provide similar symptom control as fractionated regimens but with an increased incidence of local recurrence [63–65]. A recent systematic review and meta-analysis showed no difference in sur-

vival or motor function between patients receiving short-versus long-course radiotherapy [66]. Patients experiencing local recurrence may be candidates for re-irradiation [48, 67].

While standard of care currently involves a combination of medical, surgical, and radiation therapies, recent advances in radiation oncologic underscore the increased impact of the specialty. Rades et al. showed promising results with radiation alone in treatment of myeloma [68]. Another study suggested that radiotherapy alone constitutes sufficient therapy in neurologically intact patients [69].

Stereotactic body radiation therapy (SBRT) has shown potential for treating metastatic bone disease in the spine. It has the ability to provide large doses of radiation to a highly select area providing local control of even radioresistant tumors. However its use in metastatic spinal cord compression remains limited at this time [70]. While a recent consensus study suggests that SBRT is contraindicated in cord compression [71], a new study by Ghia et al. found that it may be beneficial in select, inoperable patients [72].

Special Considerations

In spite of improved surveillance and diagnostic practices, 20% of MSCC occurs in patients without a known malignancy. A patient without a biopsy-confirmed cancer diagnosis in need of corticosteroid treatment presents a dilemma. If there is any question regarding the nature of the lesion, tissue diagnosis must be obtained without delay.

Steroids are used with curative intent in treatment of plasmacytomas, thymomas, lymphomas, multiple myeloma, and germ-cell tumors [45]. In these circumstances, corticosteroids given before tissue samples are obtained may cause regression of disease, hindering diagnosis and complicating delivery of definitive chemotherapy [73, 74]. In the absence of neurological deficit, corticosteroids may be withheld pending consultation with neurosurgery and oncologic.

Cancer patients are at increased risk of recurrent malignancy. Overall, cancer survivors have a 14% higher risk of developing a new malignancy than the general population [75]. In fact, second primary malignancies accounts for 16% of new cancer diagnoses [76].

The etiology of second primary malignancy varies. Treatment-related secondary cancers are well-documented complications of chemotherapeutic, hormonal, and radiation modalities. Familial cancer syndromes result in multiple primary cancer sites. Shared risk factors and lifestyle choices such as tobacco and alcohol use and sun exposure contribute to development of second primaries. Cancer survivors who are thought to have no evidence of disease may be experiencing either recurrent disease or the initial manifestations of a second primary process.

Prevention

Patient education is of primary importance. Lu et al. found that only 54% of patients were aware that back pain should be reported to their physician [31]. Delays in diagnosis and treatment are common and well described in the literature [26]. Patients should be instructed to call their physician within 24 hours of the development of any new or worsening back pain and should be advised to seek immediate care if they develop any neurologic symptoms.

Bisphosphonates have been shown to be effective in controlling symptoms from and prevention of skeletal metastases in breast cancer. However, they were not shown to delay development of MSCC [77, 78].

To facilitate appropriate and prompt management of MSCC, hospitals and treating physicians should develop diagnostic algorithms to minimize delays in referral to a comprehensive center for further treatment.

Prognosis

MSCC can be treated successfully; nonetheless it is associated with a poor outcome in most patients. It is evidence of an uncontrolled and aggressive disease process. Although lymphoma and myeloma patients fare better than other patients, the average lifespan after development of MSCC is less than 6 months [5, 7, 8]. Patients with limited disease and good functional status may survive for years [53].

A consensus study including 60 international experts included metastatic spinal cord compression among 11 major criteria for referral to palliative care [79]. Certainly, patients with poor functional status and those with end- or late-stage disease should be referred to palliative care for the management of symptoms [80].

Given the poor prognosis of MSCC in general, end-of-life discussions are warranted. In a retrospective study of 88 patients with MSCC, "do not resuscitate" orders were in place in only 9% of the patients during their hospital admission [81]. Improved doctor-patient communication in the ED setting will facilitate the patient's coping with future losses.

Conclusion

Metastatic spinal cord compression is an uncommon but well-established consequence of advanced cancer. It affects both survivors and patients with active disease. Patient education is a key factor in early diagnosis. It may represent the first manifestation of cancer (or a second primary cancer in the case of long-term survivors). The primary complaint is typically back pain. Neurological deficits are associated with a poor outcome. The diagnosis requires a high index of suspicion and MRI is the imaging modality of choice. Definitive treatment requires cooperative efforts by medical, surgical, and radiotherapeutic disciplines. Symptom control and maintaining (or regaining) functional status are of paramount importance. While the best predictors of outcome are ambulatory and functional status at the time of diagnosis, MSCC is generally associated with a poor prognosis. Palliative care and end-of-life issues should be considered with all patients who develop the condition.

References

- 1. Devivo MJ. Epidemiology of traumatic spinal cord injury: trends and future implications. Spinal Cord. 2012;50(5):365–72.
- Byrne TN. Spinal cord compression from epidural metastases. N Engl J Med. 1992;327(9):614–9.
- 3. Deyo RA, Rainville J, Kent DL. What can the history and physical examination tell us about low back pain? JAMA. 1992;268(6):760–5.
- Loblaw DA, Mitera G, Ford M, Laperriere NJ. A 2011 updated systematic review and clinical practice guideline for the management of malignant extradural spinal cord compression. Int J Radiat Oncol Biol Phys. 2012;84(2):312–7.
- Loblaw DA, Laperriere NJ, Mackillop WJ. A population-based study of malignant spinal cord compression in Ontario. Clin Oncol (R Coll Radiol) 2002. 2003;15(4):211–7.
- Mak KS, Lee LK, Mak RH, Wang S, Pile-Spellman J, Abrahm JL, et al. Incidence and treatment patterns in hospitalizations for malignant spinal cord compression in the United States, 1998-2006. Int J Radiat Oncol Biol Phys. 2011;80(3):824–31.
- Constans JP, de Divitiis E, Donzelli R, Spaziante R, Meder JF, Haye C. Spinal metastases with neurological manifestations. Review of 600 cases. J Neurosurg. 1983;59(1):111–8.
- Schiff D, O'Neill BP, Suman VJ. Spinal epidural metastasis as the initial manifestation of malignancy: clinical features and diagnostic approach. Neurology. 1997;49(2):452–6.
- van der Sande JJ, Boogerd W, Kröger R, Kappelle AC. Recurrent spinal epidural metastases: a prospective study with a complete follow up. J Neurol Neurosurg Psychiatry. 1999;66(5):623–7.
- Chamberlain MC, Kormanik PA. Epidural spinal cord compression: a single institution's retrospective experience. Neuron. 1999;1(2):120–3.
- Levack P, Graham J, Collie D, Grant R, Kidd J, Kunkler I, et al. Scottish Cord Compression Study Group. Don't wait for a sensory level--listen to the symptoms: a prospective audit of the delays in diagnosis of malignant cord compression. Clin Oncol (R Coll Radiol). 2002;14(6):472–80.
- Talcott JA, Stomper PC, Drislane FW, Wen PY, Block CC, Humphrey CC, et al. Assessing suspected spinal cord compression: a multidisciplinary outcomes analysis of 342 episodes. Support Care Cancer. 1999;7(1):31–8.
- Abrams HL, Spiro R, Goldstein N. Metastases in carcinoma; analysis of 1000 autopsied cases. Cancer. 1950;3(1):74–85.
- Wong DA, Fornasier VL, MacNab I. Spinal metastases: the obvious, the occult, and the impostors. Spine. 1990;15(1):1–4.
- Savage P, Sharkey R, Kua T, Schofield L, Richardson D, Panchmatia N, et al. Malignant spinal cord compression: NICE guidance, improvements and challenges. QJM. 2014;107(4):277–82.
- Schiff D. Spinal cord compression. Neurol Clin. 2003;21(1):67–86, viii.

- Schiff D, O'Neill BP, Wang CH, O'Fallon JR. Neuroimaging and treatment implications of patients with multiple epidural spinal metastases. Cancer. 1998;83(8):1593–601.
- Husband DJ, Grant KA, Romaniuk CS. MRI in the diagnosis and treatment of suspected malignant spinal cord compression. Br J Radiol. 2001;74(877):15–23.
- Arguello F, Baggs RB, Duerst RE, Johnstone L, McQueen K, Frantz CN. Pathogenesis of vertebral metastasis and epidural spinal cord compression. Cancer. 1990;65(1):98–106.
- 20. Prasad D, Schiff D. Malignant spinal-cord compression. Lancet Oncol. 2005;6(1):15–24.
- Helweg-Larsen S, Laursen H. Clinical and autopsy findings in spinal cord compression due to metastatic disease. Eur J Neurol. 1998;5(6):587–92.
- 22. Bach F, Larsen BH, Rohde K, Børgesen SE, Gjerris F, Bøge-Rasmussen T, et al. Metastatic spinal cord compression. Occurrence, symptoms, clinical presentations and prognosis in 398 patients with spinal cord compression. Acta Neurochir. 1990;107(1-2):37–43.
- Gilbert RW, Kim JH, Posner JB. Epidural spinal cord compression from metastatic tumor: diagnosis and treatment. Ann Neurol. 1978;3(1):40–51.
- Abrahm JL, Banffy MB, Harris MB. Spinal cord compression in patients with advanced metastatic cancer: "all I care about is walking and living my life". JAMA. 2008;299(8):937–46.
- Dugas AF, Lucas JM, Edlow JA. Diagnosis of spinal cord compression in nontrauma patients in the emergency department. Acad Emerg Med. 2011;18(7):719–25.
- Husband DJ. Malignant spinal cord compression: prospective study of delays in referral and treatment. BMJ. 1998;317(7150):18–21.
- 27. Shiue K, Sahgal A, Chow E, Lutz ST, Chang EL, Mayr NA, et al. Management of metastatic spinal cord compression. Expert Rev Anticancer Ther. 2010;10(5):697–708. Erratum in: Expert Rev Anticancer Ther. 2011;11(7):1148.
- 28. Hammack JE. Spinal cord disease in patients with cancer. Continuum (Minneap Minn). 2012;18(2):312–27.
- 29. National Collaborating Centre for Cancer (UK). Metastatic spinal cord compression: diagnosis and management of patients at risk of or with metastatic spinal cord compression. Cardiff (UK): National Collaborating Centre for Cancer (UK); 2008.
- Maranzano E, Bellavita R, Rossi R, De Angelis V, Frattegiani A, Bagnoli R, et al. Short-course versus split-course radiotherapy in metastatic spinal cord compression: results of a phase III, randomized, multicenter trial. J Clin Oncol. 2005;23(15):3358–65.
- Lu C, Gonzalez RG, Jolesz FA, Wen PY, Talcott JA. Suspected spinal cord compression in cancer patients: a multidisciplinary risk assessment. J Support Oncol. 2005;3(4):305–12.
- 32. Frankel HL, Hancock DO, Hyslop G, Melzak J, Michaelis LS, Ungar GH, et al. The value of postural reduction in the initial management of closed injuries of the spine with paraplegia and tetraplegia. I Paraplegia. 1969;7(3):179–92.
- Portenoy RK, Galer BS, Salamon O, Freilich M, Finkel JE, Milstein D, et al. Identification of epidural neoplasm. Radiography and bone scintigraphy in the symptomatic and asymptomatic spine. Cancer. 1989;64(11):2207–13.
- Carmody RF, Yang PJ, Seeley GW, Seeger JF, Unger EC, Johnson JE. Spinal cord compression due to metastatic disease: diagnosis with MR imaging versus myelography. Radiology. 1989;173(1):225–9.
- 35. O'Rourke T, George CB, Redmond J 3rd, Davidson H, Cornett P, Fill WL, et al. Spinal computed tomography and computed tomographic metrizamide myelography in the early diagnosis of metastatic disease. J Clin Oncol. 1986;4(4):576–83.
- Bilsky MH, Laufer I, Fourney DR, Groff M, Schmidt MH, Varga PP, et al. Reliability analysis of the epidural spinal cord compression scale. J Neurosurg Spine. 2010;13(3):324–8.

- Bilsky M, Smith M. Surgical approach to epidural spinal cord compression. Hematol Oncol Clin North Am. 2006;20(6):1307–17. https://doi.org/10.1016/j.hoc.2006.09.009.
- Arana E, Kovacs FM, Royuela A, Asenjo B, Pérez-Ramírez Ú, Zamora J. Spanish back pain research network task force for the improvement of inter-disciplinary management of spinal metastasis. Agreement in metastatic spinal cord compression. J Natl Compr Cancer Netw. 2016;14(1):70–6.
- Lee SH, Grant R, Kennedy C, Kilbride L. Positioning and spinal bracing for pain relief in metastatic spinal cord compression in adults. Cochrane Database Syst Rev. 2015;2015(9):Cd007609.
- Paniagua-Collado M, Cauli O. Non-pharmacological interventions in patients with spinal cord compression: a systematic review. J Neuro-Oncol. 2018;136(3):423–34.
- Sørensen S, Helweg-Larsen S, Mouridsen H, Hansen HH. Effect of high-dose dexamethasone in carcinomatous metastatic spinal cord compression treated with radiotherapy: a randomised trial. Eur J Cancer. 1994;30a(1):22–7.
- 42. Heimdal K, Hirschberg H, Slettebø H, Watne K, Nome O. High incidence of serious side effects of high-dose dexamethasone treatment in patients with epidural spinal cord compression. J Neuro-Oncol. 1992;12(2):141–4.
- 43. L'espérance S, Vincent F, Gaudreault M, Ouellet JA, Li M, Tosikyan A. Goulet S; Comité de l'évolution des pratiques en oncologie. Treatment of metastatic spinal cord compression: cepo review and clinical recommendations. Curr Oncol. 2012;19(6):e478–90.
- Kumar A, Weber MH, Gokaslan Z, Wolinsky JP, Schmidt M, Rhines L, et al. Metastatic spinal cord compression and steroid treatment: a systematic review. Clin Spine Surg. 2017;30(4):156–63.
- Posner JB, Howieson J, Cvitkovic E. "Disappearing" spinal cord compression: oncolytic effect of glucocorticoids (and other chemotherapeutic agents) on epidural metastases. Ann Neurol. 1977;2(5):409–13.
- 46. Flouzat-Lachaniette CH, Allain J, Roudot-Thoraval F, Poignard A. Treatment of spinal epidural compression due to hematological malignancies: a single institution's retrospective experience. Eur Spine J. 2013;22(3):548–55.
- 47. Gokaraju K, Butler JS, Benton A, Suarez-Huerta ML, Selvadurai S, Molloy S. Multiple myeloma presenting with acute bony spinal cord compression and mechanical instability successfully managed nonoperatively. Spine J. 2016;16(8):e567–70.
- Rades D, Abrahm JL. The role of radiotherapy for metastatic epidural spinal cord compression. Nat Rev Clin Oncol. 2010;7(10):590–8.
- Ropper AE, Ropper A. Acute spinal cord compression. N Engl J Med. 2017;376(14):1358–69.
- 50. Fisher CG, DiPaola CP, Ryken TC, Bilsky MH, Shaffrey CI, Berven SH, et al. A novel classification system for spinal instability in neoplastic disease: an evidence-based approach and expert consensus from the Spine Oncology Study Group. Spine (Phila Pa 1976). 2010;35(22):E1221–9. https://doi.org/10.1097/ BRS.0b013e3181e16ae2.
- Akram H, Allibone J. Spinal surgery for palliation in malignant spinal cord compression. Clin Oncol (R Coll Radiol). 2010;22(9):792–800.
- 52. Younsi A, Riemann L, Scherer M, Unterberg A, Zweckberger K. Impact of decompressive laminectomy on the functional outcome of patients with metastatic spinal cord compression and neurological impairment. Clin Exp Metastas. 2020;37(2):377–90.
- Sundaresan N, Sachdev VP, Holland JF, et al. Surgical treatment of spinal cord compression from epidural metastasis. J Clin Oncol. 1995;13(9):2330–5.
- 54. Vega RA, Traylor JI, Habib A, Rhines LD, Tatsui CE, Rao G. Minimally invasive separation surgery for metastases in the vertebral column: a technical report. Oper Neurosurg Hagerstown. 2020;18(6):606–13.

- Chen F, Xia YH, Cao WZ, et al. Percutaneous kyphoplasty for the treatment of spinal metastases. Oncol Lett. 2016;11(3):1799–806.
- 56. Kwan MK, Lee CK, Chan CY. Minimally invasive spinal stabilization using fluoroscopic-guided percutaneous screws as a form of palliative surgery in patients with spinal metastasis. Asian Spine J. 2016;10(1):99–110.
- 57. Chang CW, Fu TS, Lin DY, Lai PL, Chiu PY, Kao FC, et al. Percutaneous balloon kyphoplasty and short instrumentation compared with traditional long instrumentation for thoracolumbar metastatic spinal cord compression. World Neurosurg. 2019;130:e640–7.
- 58. Kim JM, Losina E, Bono CM, Schoenfeld AJ, Collins JE, Katz JN, Harris MB. Clinical outcome of metastatic spinal cord compression treated with surgical excision ± radiation versus radiation therapy alone: a systematic review of literature. Spine (Phila Pa 1976). 2012;37(1):78–84.
- Patchell RA, Tibbs PA, Regine WF, Payne R, Saris S, Kryscio RJ, et al. Direct decompressive surgical resection in the treatment of spinal cord compression caused by metastatic cancer: a randomised trial. Lancet. 2005;366(9486):643–8.
- Prewett S, Venkitaraman R. Metastatic spinal cord compression: review of the evidence for a radiotherapy dose fractionation schedule. Clin Oncol (R Coll Radiol). 2010;22(3):222–30.
- 61. Steenland E, Leer JW, van Houwelingen H, Post WJ, van den Hout WB, Kievit J, et al. The effect of a single fraction compared to multiple fractions on painful bone metastases: a global analysis of the Dutch Bone Metastasis Study. Radiother Oncol. 1999;52(2):101–9.
- 62. Rades D, Lange M, Veninga T, Stalpers LJ, Bajrovic A, Adamietz IA, et al. Final results of a prospective study comparing the local control of short-course and long-course radiotherapy for meta-static spinal cord compression. Int J Radiat Oncol Biol Phys. 2011;79(2):524–30.
- 63. van den Hout WB, van der Linden YM, Steenland E, Wiggenraad RG, Kievit J, de Haes H, Leer JW. Single- versus multiple-fraction radiotherapy in patients with painful bone metastases: costutility analysis based on a randomized trial. J Natl Cancer Inst. 2003;95(3):222–9.
- 64. Rades D, Fehlauer F, Schulte R, Veninga T, Stalpers LJ, Basic H, et al. Prognostic factors for local control and survival after radiotherapy of metastatic spinal cord compression. J Clin Oncol. 2006;24(21):3388–93.
- 65. Rich SE, Chow R, Raman S, Liang Zeng K, Lutz S, Lam H, et al. Update of the systematic review of palliative radiation therapy fractionation for bone metastases. Radiother Oncol. 2018;126(3):547–57.
- 66. Qu S, Meng HL, Liang ZG, Zhu XD, Li L, Chen LX, Zhou ZR. Comparison of short-course radiotherapy versus long-course radiotherapy for treatment of metastatic spinal cord compression: a systematic review and meta-analysis. Medicine (Baltimore). 2015;94(43):e1843.
- 67. Mancosu P, Navarria P, Bignardi M, Cozzi L, Fogliata A, Lattuada P, et al. Re-irradiation of metastatic spinal cord compression: a feasibility study by volumetric-modulated arc radiotherapy for

in-field recurrence creating a dosimetric hole on the central canal. Radiother Oncol. 2010;94(1):67–70.

- Rades D, Conde-Moreno AJ, Cacicedo J, Segedin B, Rudat V, Schild SE. Excellent outcomes after radiotherapy alone for malignant spinal cord compression from myeloma. Radiol Oncol. 2016;50(3):337–40.
- Maranzano E, Latini P, Beneventi S, Perruci E, Panizza BM, Aristei C, et al. Radiotherapy without steroids in selected metastatic spinal cord compression patients. A phase II trial. Am J Clin Oncol. 1996;19(2):179–83.
- Barzilai O, Boriani S, Fisher CG, Sahgal A, Verlaan JJ, Gokaslan ZL, et al. Essential concepts for the management of metastatic spine disease: what the surgeon should know and practice. Global Spine J. 2019;9(1 Suppl):98S–107S.
- Redmond KJ, Lo SS, Soltys SG, Yamada Y, Barani IJ, Brown PD, et al. Consensus guidelines for postoperative stereotactic body radiation therapy for spinal metastases: results of an international survey. J Neurosurg Spine. 2017;26(3):299–306.
- 72. Ghia AJ, Guha-Thakurta N, Hess K, Yang JN, Settle SH, Sharpe HJ, et al. Phase 1 study of spinal cord constraint relaxation with single session spine stereotactic radiosurgery in the primary management of patients with inoperable, previously unirradiated metastatic epidural spinal cord compression. Int J Radiat Oncol Biol Phys. 2018;102(5):1481–8.
- Kan E, Levi I, Benharroch D. Alterations in the primary diagnosis of lymphomas pretreated with corticosteroid agents. Leuk Lymphoma. 2011;52(3):425–8.
- Borenstein SH, Gerstle T, Malkin D, Thorner P, Filler RM. The effects of prebiopsy corticosteroid treatment on the diagnosis of mediastinal lymphoma. J Pediatr Surg. 2000;35(6):973–6.
- 75. Curtis RE, Freedman DM, Ron E, Ries LAG, Hacker DG, Edwards BK, et al., editors. New malignancies among cancer survivors: SEER cancer registries, 1973-2000. Bethesda: National Cancer Institute. NIH Publ. No. 05-5302; 2006.
- Travis LB. The epidemiology of second primary cancers. Cancer Epidemiol Biomark Prev. 2006;15(11):2020–6.
- Ross JR, Saunders Y, Edmonds PM, Patel S, Broadley KE, Johnston SR. Systematic review of role of bisphosphonates on skeletal morbidity in metastatic cancer. BMJ. 2003;327(7413):469.
- Wang Z, Qiao D, Lu Y, Curtis D, Wen X, Yao Y, Zhao H. Systematic literature review and network meta-analysis comparing bonetargeted agents for the prevention of skeletal-related events in cancer patients with bone metastasis. Oncologist. 2015;20(4):440–9.
- Hui D, Mori M, Watanabe SM, Caraceni A, Strasser F, Saarto T, et al. Referral criteria for outpatient specialty palliative cancer care: an international consensus. Lancet Oncol. 2016;17(12):e552–9.
- Rades D, Hueppe M, Schild SE. A score to identify patients with metastatic spinal cord compression who may be candidates for best supportive care. Cancer. 2013;119(4):897–903.
- Guo Y, Palmer JL, Bianty J, Konzen B, Shin K, Bruera E. Advance directives and do-not-resuscitate orders in patients with cancer with metastatic spinal cord compression: advanced care planning implications. J Palliat Med. 2010;13(5):513–7.

Head and Neck



Ray Y. Wang, Eugene L. Son, Clifton D. Fuller, and Neil D. Gross

Introduction

Head and neck cancer is the fifth most common cancer worldwide with almost 53,000 people diagnosed each year in the United States [1]. These include malignancies of the upper aerodigestive tract (UADT), which begins at the lips and nose and extends to the mucosal surfaces of the cervical trachea and esophagus. The UADT includes the oral cavity, oropharynx, hypopharynx, larynx, nasal cavities, and paranasal sinuses. The most common malignancy of the UADT is squamous cell carcinoma of the head and neck (SCCHN), making up over 97% [2]. For simplicity, SCCHN will be the primary focus of this chapter. Other pathologies in the head and neck including carcinoma of the major and minor salivary glands, thyroid, and skin will not be covered.

Patients with SCCHN may present to the emergency department (ED) with a variety of complaints at any point in the course of their disease, from initial diagnosis to years after completion of therapy. Patients presenting on an emergency basis represent 7% of all referrals with dysphagia, stridor, and neck mass being the most common chief complaints [3]. Also typical, pain and gastrointestinal complaints were the most common presenting symptoms among patients with SCCHN in a population-based study from Taiwan [4]. The symptom profile of patients presenting may vary over their treatment course, although pain is a consistent symptom [5].

R. Y. Wang

Department of Otolaryngology-Head and Neck Surgery, Baylor College of Medicine, Houston, TX, USA

E. L. Son

Department of Head & Neck Surgery, Kaiser Permanente, Fontana, CA, USA

C. D. Fuller

Department of Radiation Oncology, The University of Texas MD Anderson Cancer Center, Houston, TX, USA

N. D. Gross (🖂)

In one prospective cohort study, patients presenting during or soon after treatment presented most frequently with gastrointestinal complaints while those presenting >6 months after treatment initiation were most likely to present with pain [5].

Uncontrolled SCCHN can result in life-threatening emergencies, principally from compromise of the airway and/or bleeding. Patients with undiagnosed SCCHN may present to the ED in distress because the SCC of the base of tongue and larynx can grow to be large with relatively few symptoms [6]. In this chapter, we discuss acute management of the airway in patients with SCCHN including in the context of COVID-19. We also review the management of emergent bleeding in patients with SCCHN. Finally, we cover common complications from the treatment of SCCHN and end with clinical pearls for acute management of SCCHN in the ED.

Case Discussion

A 59-year-old man with a history of recurrent oropharyngeal squamous cell carcinoma presented to the ED for evaluation of bleeding from his neck and mouth. A few hours prior to presentation, he suddenly developed severe bleeding ("several cupsful") externally from his tumor as well as from his oral and nasal cavity. This resolved spontaneously prior to arrival in the emergency room.

He has a complex oncologic history including squamous cell carcinoma of the base of tongue initially treated with concurrent chemoradiation approximately 2 years prior to the presentation followed by partial glossectomy, neck dissection, and free flap reconstruction for recurrence a few months later. He again was noted to have recurrence with extensive progression involving the neck soft tissues with multiple pharyngocutaneous fistulae.

On examination, he was afebrile, tachycardic (HR 110s) but normotensive. He was breathing well on room air without stridor though with audible pharyngeal secretions. He had extensive tumor involvement of the anterior neck with

Division of Surgery, Department of Head and Neck Surgery, The University of MD Anderson Cancer Center, Houston, TX, USA e-mail: ngross@mdanderson.org

K. H. Todd et al. (eds.), Oncologic Emergency Medicine, https://doi.org/10.1007/978-3-030-67123-5_19

multiple fistulae with dark clot but no active bleeding. He also had post-treatment sequelae including radiation fibrosis with limited neck range of motion and trismus from his prior surgery and reconstruction. Flexible fiberoptic nasopharyngoscopy demonstrated a large blood clot at the left base of tongue at the site of his recurrence with marked supraglottic edema.

Given the potential for acute airway compromise due to bleeding from his pharyngeal tumor, he was taken emergently to the operating room where he underwent awake transoral fiberoptic intubation followed by tracheostomy. After securing the airway, a CT angiogram of the head and neck was obtained which did not demonstrate obvious extravasation. He was then taken by interventional radiology for angiography and a pseudoaneurysm of the left facial artery was noted. This was managed via selective arterial embolization. The patient demonstrated no additional bleeding after the procedure.

Airway Management

Airway obstruction due to SCCHN affects up to 80,000 patients annually, with most patients presenting to the ED for acute care [7]. Ideally, planning for airway management in patients with SCCHN should occur well before an emergency. This may include prophylactic tracheostomy. In addition to acute airway obstruction, SCCHN patients have increased rates of aspiration pneumonia because of dysphagia and difficulty handling even normal oral secretions [8]. The 1-year and 5-year incidence of clinically meaningful aspiration in SCCHN patients is 15.8% and 23.8%, respectively, with 84% of these patients requiring hospitalization [9]. For the purposes of this chapter, we have divided the airway into *unsecure* and *secure*.

Unsecured Airway

Undiagnosed SCCHN can lead to signs of obstruction before causing other symptoms such as pain [10]. Metastatic cervical adenopathy can also cause obstructive lymphedema and direct extrinsic compression of the airway. Patients with undiagnosed SCCHN may present with orthopnea, hoarseness, dysphagia, odynophagia, and hemoptysis [10]. Stridor and/or drooling can be signs of an unsecured airway.

In stable patients, a contrasted computed tomography (CT) of the head, neck, and chest can be performed quickly to assess the location and extent of the obstruction [10]. This imaging study provides important staging information if malignancy is confirmed. Flexible fiberoptic laryngoscopy (FFL) is a critical tool for assessing the airway and is the most direct method to evaluate for impending obstruction

Fig. 19.1 Flexible fiberoptic laryngoscopy demonstrating normal larynx

[10] (Fig. 19.1). Skilled management of the FFL is important for evaluating a potentially tenuous airway. Topical administration of local anesthetic such as lidocaine and topical decongestants (oxymetazoline or phenylephrine) to the nasal cavities may make FFL more tolerable and potentially safer (see Fig. 19.1). Management of a suspected unsecure airway starts with optimal positioning to make the patient as comfortable as possible. Administration of supplemental oxygen may be helpful but should be used with caution in patients with uncompensated chronic obstructive pulmonary disease (COPD). Administrating nebulizers and steroids are unlikely to improve the airway but may be useful in temporizing the patient. The use of Heliox, a mixture of helium and oxygen, has been described in the acute management of patients with an unsecure airway. The decreased viscosity of Heliox can temporarily improve airflow and reduce stridor. Because it is an inert gas, it can assist in temporizing an unsecure airway, but it may not be readily available in the ED [11].

SCCHN patients with acute respiratory failure from an unsecure airway should be managed expeditiously. Transoral intubation is preferred if FFL predicts that direct visualization of the endolarynx can be achieved safely. The use of laryngeal mask anesthesia (LMA) is not recommended in these cases because of likely distortion of the normal anatomy. In cases where oral intubation is not possible (e.g., obstructing mass, trismus), awake fiberoptic transoral or nasotracheal intubation is usually the next best option. Nasotracheal intubation is preferred over awake fiberoptic oral intubation to decrease patient gagging but requires a unique skill set and is probably best performed by experienced anesthesiologists or otolaryngologists. Successful intubation can avoid an emergent surgical airway and allow a controlled environment for formal tracheostomy [6]. SCCHN involving the larynx is most likely to result in an unsecured airway (Fig. 19.2). Intubation of a patient with





Fig. 19.2 Flexible fiberoptic laryngoscopy demonstrating obstructing mass of the larynx. Note loss of visualization of the vocal cords and the markedly decreased diameter of the airway

locally advanced SCC of the larynx can be hazardous because of the distorted anatomy and risk of bleeding during laryngoscopy. In these cases, the airway is best managed in the operating room by an experienced anesthesiologist and/or otolaryngologist.

If attempted intubation is deemed unsafe or unsuccessful, then a surgical airway via cricothyroidotomy or tracheostomy should be performed. Awake tracheostomy can be performed under local anesthesia in select SCCHN patients with an unsecure airway. This is best performed in the operating room. If an urgent airway is needed (e.g., acute obstruction) then a "slash" cricothyroidotomy should be performed. A vertical incision is usually advised in this setting, as a midline dissection is critical to minimize bleeding from the paired paramedian anterior jugular veins and allowing for identification of the airway across a vertical range. In cases where the trachea is deviated, a needle on a saline-filled syringe with negative pressure can be used to locate the trachea by visualization of air bubbles [12]. Transtracheal catheterization has also been described if cricothyroidotomy or tracheostomy cannot be performed [13].

Some head and neck cancers, particularly thyroid cancers, can cause paralysis to one or both of the vocal cords via direct involvement of the recurrent laryngeal nerves (Fig. 19.3). This is in contrast to large goiters that can distort the airway over time but rarely cause respiratory distress (Fig. 19.4). Acute bilateral vocal fold paralysis, either by direct tumor involvement or iatrogenic after thyroid surgery, can cause respiratory distress marked by stridor. These patients typically require intubation and subsequent tracheostomy if possible. Anaplastic thyroid cancer is the prototypic malignancy to cause acute airway obstruction by direct tracheal involvement, recurrent laryngeal nerve involvement, or both. Fortunately, anaplastic thyroid carcinoma comprises



Fig. 19.3 Axial computed tomography (CT) of the neck demonstrating a left-sided thyroid mass with invasion into the cricoid cartilage. This patient presented with a paralyzed ipsilateral vocal cord secondary to involvement of the recurrent laryngeal nerve



Fig. 19.4 Axial computed tomography (CT) of the neck demonstrating a large goiter with narrowing of the airway but no obstruction

only 1.7% of all thyroid cancers [14]. Management of the airway in patients with anaplastic thyroid cancer is complex and controversial particularly given the dismal prognosis associated with the disease [15]. For this reason, current American Thyroid Association guidelines recommend against elective tracheostomy [14].

Secured Airway

The airway management of SCCHN patients with a tracheostomy or laryngectomy stoma requires some familiarity with changes to the anatomy with these procedures. Most clinicians are familiar with tracheostomy patients whereby the oral cavity, oropharynx, and larynx are bypassed by a tube directly into the trachea. In contrast, laryngectomy patients are obligate "neck breathers" with no remaining connection between the mouth and trachea.

As with all patients presenting with respiratory distress, conservative measures should always be initiated including oxygen administration. However, nasal cannula or facemask administration may be ineffective if the patient breathes through a surgically created stoma in the neck. Oxygen can be applied to both the face and stoma for tracheostomy patients but only the stoma for laryngectomy patients. Fiberoptic tracheoscopy can be a valuable tool to rule out a proximal obstruction or false passage of the tracheostomy tube. The scope can be introduced inside the stoma or tube in place to visualize the carina and proximal mainstem bronchi.

SCCHN patients with a tracheostomy are at risk of lifethreatening complications, including bleeding and tube dislodgement with airway obstruction and death [16]. There are known late complications of tracheostomy in up to 65% of patients including possible granulation tissue formation, tracheomalacia, tracheoinnominate fistula (TIF), tracheoesophageal fistula, pneumonia, and aspiration [17]. These potential complications from tracheostomy tubes are important to recognize in the acute setting.

Patients can present to the ED with airway obstruction despite having a tracheostomy tube. If a tracheostomy tube becomes dislodged, then every effort should be made to replace the tube as the stoma can close substantially in a matter of hours. If the original tracheostomy tube is too large for the tracheostomy stoma at the time of replacement, then the tract can be dilated with a nasal speculum or a smaller tube can be inserted. An endotracheal tube can even be used temporarily to secure the airway if needed. However, it is important to keep the cuff visible near the stoma to avoid a mainstem bronchus intubation.

Mucous plugging of a tracheostomy tube can cause acute airway obstruction and death. For this reason, most commercially available tracheostomy tubes have an interchangeable inner cannula. Patients need appropriate humidification of air and frequent suctioning of the tube to prevent mucous build-up. Applying small amounts of saline and suctioning with a soft flexible catheter can soften and remove hardened mucous.

There are special considerations regarding the emergency management of laryngectomy patients. A survey of members of the National Association of Laryngectomy Clubs in the United Kingdom underscored concerns regarding the quality of care they receive in the emergency setting [18]. It is incumbent upon emergency physicians to be familiar with the anatomy of patients after laryngectomy and the common complications that can occur. For example, it is important to realize that either a standard tracheostomy tube or more customized laryngectomy tube may be used for comfort or to stent the stoma. In the superior posterior



Fig. 19.5 Laryngectomy stoma with blue tracheoesophageal puncture (TEP) voice prosthesis in place

wall of the trachea, a tracheoesophageal puncture (TEP) device may be present which provides a one-way valve for air to flow from the trachea into the esophagus for speech (Fig. 19.5). If a patient requires mask ventilation, then the laryngectomy patient should be intubated through the stoma either with a cuffed endotracheal or tracheostomy tube past the TEP, if present. The rate of 30-day unplanned readmission independent of complications following laryngectomy ranges from 26.5% to 42% [19]. Most of these patients present for stomal or TEP complications [19]. Non-humidified air entering the proximal trachea can cause thickened respiratory secretions that are difficult to clear. These secretions can dry into dense circumferentially crusts which can obstruct the airway. Post-laryngectomy patients also frequently come to the ED for TEP dislodgement. When this happens, a chest film must be taken to rule out aspiration of the device as aspiration has been reported to occur in up to 13% of patients [20]. If the patient comes with a dislodged TEP device in hand, replacement can be difficult without extensive experience and specialized tools. Replacement of the TEP is usually done by a speech and language pathologist or otolaryngologist. If replacement is not an immediate option, then placement of a temporary red rubber catheter through the TEP site can help prevent aspiration and closure of the puncture itself.

Considerations During the SARS-CoV-2 (COVID-19) Pandemic

The SARS-CoV-2 (COVID-19) pandemic introduced significant new challenges for providers caring for head and neck cancer patients. Given the anatomic location of SCCHN, care for these patients places providers at high risk for transmission due to the need to examine the UADT. Furthermore, cancer patients may be at increased risk for severe complications related to COVID-19, including ICU admission, need for mechanical ventilation, or death [21].

Emergency airway management in head and neck cancer patients presents several unique challenges as concerns for COVID-19 transmission disrupt the typical workflow of care. Ideally, all patients undergoing aerosol-generating procedures would have pre-procedure testing for COVID-19, but in emergency circumstances, this may not be possible. As such, these patients must be treated as persons under investigation (PUI) for COVID-19. Precautions taken to reduce potential exposure of healthcare providers must be balanced with the need to provide expeditious care for these critically ill patients. FFL, a mainstay of airway evaluation in head and neck patients, is considered an aerosolgenerating procedure and appropriate personal protective equipment (PPE) including an N95 mask or powered air purifying respiratory (PAPR), eye protection, and gloves is recommended by the American Academy of Otolaryngology-Head and Neck Surgery [22]. Steps should be taken to reduce the number of providers exposed during the procedure, such as recording the exam or having a senior member of the team perform the exam alone. Airway support measures, such as high-flow nasal cannula or non-invasive positive pressure ventilation, are also at high risk for aerosolization of respiratory secretions and should be used with caution [23].

Awake tracheostomy is particularly high risk for transmission of COVID-19 given the high likelihood of coughing during the procedure, inability to control ventilation when entering the airway, and respiratory distress of the patient during the procedure [24]. All personnel involved in these procedures should wear full PPE including N95 masks or PAPRs. Protocols for emergency airway management during the COVID-19 pandemic are not standardized and may not be tailored specifically for the head and neck cancer population. When possible, fiberoptic intubation after application of sufficient topical anesthetic should be performed as it allows for controlled entry of the airway and the ability to hold ventilation, thus reducing the aerosolization of respiratory secretions. In patients with difficult anatomy or in emergent circumstances this algorithm may not be feasible. In these cases, the providers should proceed directly to awake tracheostomy and take all possible precautions to avoid unnecessary exposure of healthcare personnel to respiratory secretions. General recommendations from the New York Head and Neck Society regarding tracheostomy for COVID-19 patients included performing tracheostomy in a negative pressure room (e.g., in the intensive care unit or a designated pod within the operating room), preoxygenation and holding ventilation when entering the airway, avoiding monopolar cautery or harmonic technology and use of cold instrumentation, minimizing the use of suction, and potentially utilizing percutaneous tracheostomy when feasible [25]. Alternative measures such as securing a face mask with 251

a seal around the mouth and nose rather than the standard nasal cannula may reduce aerosolization during awake tracheostomy [24].

Bleeding Management

Patients with head and neck cancer, most notably SCCHN, can develop life-threatening bleeding. There is a rich vascular supply to the head and neck region. Bleeding can occur from direct tumor involvement and/or as a side effect of therapy [26]. The most common cause of bleeding is poor wound healing after surgery or radiation. The initial management of bleeding in head and neck cancer patients is no different than the general population. The patient must first be stabilized. The ABCs (Airway, Breathing, and Circulation) of shock trauma should be addressed. Two large-bore intravenous lines should be obtained. Warmed isotonic electrolyte solutions such as lactated Ringer's solution or normal saline should be administered in a bolus fashion. Transfusion of packed red blood cells (pRBC) should be considered. When administering large amounts of pRBC, calcium supplementation should be considered as there are chelating agents of calcium in these blood products [27]. Laboratory studies including complete blood count (CBC), prothrombin time (PT), activated partial thromboplastin time (aPTT), and other coagulation labs should be performed and corrected as needed to assist in hemostasis.

Acute Arterial Bleeding

Acute arterial bleeding from the mouth or neck can occur after treatment of SCCHN. Surgery can strip the vascular supply of the arterial wall. Radiation therapy can cause obliteration of the vasa vasorum, premature atherosclerosis, adventitial fibrosis, and fragmentation of tunica media elastic fibers leading to weakening of the arterial wall [28]. Patients with head and neck cancer have other factors that contribute to poor wound healing including nutritional compromise, poor tissue perfusion, soft tissue exposure to salivary enzymes, and infections [29].

Major arterial bleeding is often preceded by a sentinel bleed, usually from a pseudoaneurysm, that can be profuse but self-limited [29]. A spontaneous cessation of brisk bleeding in a patient with SCCHN can give the emergency physician a false sense of safety. Immediate diagnostic workup followed by treatment should be obtained to prevent a catastrophic bleed. CT imaging may show irregular thickening of the arterial wall of major vessels [30]. If the patient is stabilized, CT angiography (CTA) can be an effective screening tool for locating the site of hemorrhage and can also assist in



Fig. 19.6 Angiogram of the right internal maxillary artery in a patient with epistaxis from a sinonasal cancer before (a) and after (b) embolization

procedures performed by the intervention neuroradiologist [29]. Prophylactic treatment of the diseased vessel can prevent catastrophic events (Fig. 19.6).

Direct pressure is key to temporarily controlling any acute bleeding. In the head and neck, there is the additional challenge of managing the airway. If the bleeding is origination from the oral cavity or oropharynx, then the airway must be secured before effective pressure can be applied. With the airway secured, bleeding from the mouth can generally be temporized with a throat pack. If a tracheostomy tube is present, the cuff should be immediately inflated to prevent aspiration of blood into the lower airway and lungs. If the tracheostomy tube does not have a cuff, then it should be replaced with a cuffed tracheostomy tube. If a significant amount of blood becomes static in the trachea and the bronchial tree, then clots will cause total obstruction of the airway. This risk is much lower in laryngectomy patients since the oral cavity and oropharynx have no connection to the airway. Even so, it is advisable to intubate the trachea for protection in these patients during any significant bleeding episode. Once the airway is secure, the underlying cause of bleeding can be investigated and managed by a head and neck surgeon and/or interventional neuroradiologist.

Carotid Blowout

The most feared bleeding complication from SCCHN is carotid rupture (aka carotid blowout). Without immediate, aggressive intervention, carotid blowout is uniformly fatal. Before interventional angiography, hemostasis for carotid artery rupture was obtained via an open surgical approach. This was associated with 60% neurological morbidity and 40% mortality. More recently, endovascular stenting of a carotid artery rupture has shown >80% survival and far fewer neurological events. However, endovascular stent placement without further treatment has a recurrent bleeding up to 26% [29]. So, the patients who are successfully resuscitated and stented should undergo subsequent definitive surgical management (e.g., overlying soft tissue flap reconstruction) to prevent further bleeding episodes [31]. In dire cases, the ipsilateral carotid artery can be permanently occluded, albeit with at least a 15–20% risk of delayed cerebral complications [32].

Internal Jugular Vein Bleeding

Internal jugular vein bleeding, although rarer than carotid artery bleeding, can occur after treatment of SCCHN. These are typically less severe, characterized by multiple episodes, and aggravated by coughing [32]. Internal jugular vein bleeding is almost uniformly associated with a pharyngocutaneous fistula [32]. The treatment is a surgical exploration of the wound and ligation.

Tracheoinnominate Fistula

A tracheoinnominate fistula (TIF) is a connection from the trachea to the innominate artery. It is a rare complication after tracheostomy placement ranging from 0.1% to 1% in incidence and usually occurring between postoperative days 7 and 14 [31]. There are a number of factors that can predispose a tracheostomy patient to this complication including lower placed tracheotomies, over-inflated cuffs causing ero-

sion of the trachea, and anomalies of the innominate or other large caliber arteries [33]. Before this catastrophic event, there may be an ominous sign of milder pulsating bleeding from the tracheostomy (sentinel bleed) [33].

TIF has a high rate of mortality as it causes rapid exsanguination in combination with aspiration of large amounts of blood. The mortality rate approaches 100%, even when surgical intervention is taken [34]. Definitive management of a TIF requires a sternotomy and vascular repair in the operating room. Placing direct pressure against the anterior tracheal wall can temporize the bleeding. This can be done either digitally with a finger or by placing a cuffed tube, creating a temporary tamponade [33]. Endovascular embolization or placement of a stent graft of the innominate artery has also been reported [31].

Epistaxis

Epistaxis is a common reason for presentation to the ED. Management of epistaxis from head and neck malignancy or after surgery is different from ordinary epistaxis and merits special considerations. A typical case of epistaxis can usually be watched and stopped with external digital pressure and lubrication of the nasal mucosa with nasal saline spray and antibiotic ointment. Epistaxis in the setting of a sinonasal cancer can be more serious and should be handled more aggressively.

With active epistaxis, visualization with anterior rhinoscopy or endoscopy may not be possible depending on the volume of bleeding. The patient should be sitting up and positioned leaning forward to allow the bleeding to exit the nares rather than into the pharynx, where it may cause airway compromise. If the bleeding is severe, the airway may need to be secured with intubation. Packing is often required to tamponade uncontrolled epistaxis. There are dissolvable packing materials, such as cellulose polymers (Surgicel®) and porcine skin gelatin (Gelfoam[®]), and non-dissolvable packing materials, such as cross-linked polyvinyl alcohol (Merocel®) and balloon packs (Rapid Rhino®). A headlight, nasal speculum, and bayonet forceps should be used for placement of packing. Other products such as topical thrombin components (Floseal[®]) can be topically placed inside the nose to aid in hemostasis.

Given the high viral load of SARS-CoV-2 in the nasal cavity and nasopharynx, providers should proceed with caution when managing patients with known COVID-19 or unknown infection status [35]. When possible, testing for COVID-19 should be obtained prior to intervention for epistaxis. Nasal packing and other procedures to manage epistaxis are high-risk aerosol-generating procedures and appropriate PPE should be utilized, including N95 masks or PAPRs, eye protection, gown, and gloves. Noninvasive measures such as digital compression or medical management should be attempted for mild epistaxis before more invasive measures are undertaken. Absorbable packing should be considered to reduce the need for further procedures to remove packing.

Management of Treatment Complications

Surgical Complications

There are risks of complications after surgery for SCCHN beyond those already covered in this chapter (airway and bleeding). The most common complications after surgery are edema and seroma formation.

After neck dissection (cervical lymphadenectomy), it is common for patients to develop lymphedema of the lower face and neck. Patients who receive adjuvant treatment including radiation therapy and chemotherapy often have lymphedema after treatment. This most commonly presents in the neck and submental region with pitting and non-pitting edema [36]. Seroma is a collection of sterile, straw-colored serous fluid in a dead space of the surgical field and most commonly happens after neck dissection and thyroidectomy. A seroma can easily be mistaken for an abscess if the skin is red and tender. A white blood cell (WBC) may not be helpful as this can be elevated after surgery without infection. A key difference is that there are typically no clinical signs of severe infection (e.g., fever, sepsis) with seroma. If there is uncertainty, then sterile needle aspiration of the fluid can be diagnostic. Seromas typically do not require emergent treatment.

In contrast to seroma, a hematoma is a collection of blood within the surgical bed and can occur in up to 4% of all major head and neck cancer surgeries [37]. A hematoma can be distinguished from seroma by the presence of bruising and turgor of the overlying skin (Fig. 19.7). An expanding hematoma of the neck should be recognized as an emergency because of the potential for airway compression. The treatment of an expanding hematoma is evacuation of the hematoma and control of any bleeding vessels. This is best performed sterilely in the operating room to reduce the risk of infection but may be necessary at the bedside if the patient develops an unstable airway.

A chyle leak is an uncommon complication that can occur after surgery in the low neck. A chyle leak can present in a similar manner to a seroma. The defining difference is that chyle has a characteristic milky color with normal dietary intake and can have inflammatory effects. Chyle leaks present soon after major neck surgery often when drains are still in place. Most chyle leaks occur on the left due to the pres-



Fig. 19.7 Hematoma of the upper neck 24 h after neck dissection. Note ecchymosis of the overlying skin

ence of the thoracic duct emptying in the left subclavian vein near the internal jugular vein. Needle aspiration of a suspected chyle leak will be unproductive. Most chyle leaks can be managed conservatively with a no-fat diet and continuation of a drain.

Salivary fistula is a complication distinct to surgery of the head and neck involving the parotid gland. A salivary fistula can present like a seroma but is treated differently. There may be other signs of infection including erythema, turbid fluid in the drain, purulence, or wound breakdown. Diagnostic needle aspiration can be performed but the fluid should be tested for amylase which would be unique to saliva. Pharyngocutaneous fistula, a connection from the pharynx to the skin, can occur after major head and neck surgery, particularly in patients who have had prior treatment with radiation [38]. Treatment typically includes incision, drainage, and packing of the wound.

Radiation Therapy Complications

Radiation therapy (RT) is a commonly used method of treatment for head and neck cancer, particularly SCCHN. RT-related toxicities in the head and neck include erythema, ulceration, xerostomia, lymphedema, fibrosis, and osteoradionecrosis (ORN) of the jaw or temporal bone. RT has early and late effects on normal tissue. The early effects are caused by DNA damage and reactive oxygen species formation with resultant cell death to ciliated epithelium, blood vessels, and secretory glands [39]. The late effects are caused by ischemia from microvascular damage and fibrosis, which cause tissue edema, erythema, hemorrhage, and thickened secretions [39]. The combination of chemotherapy and radiation therapy (CRT) in the head and neck causes a synergistic effect on cancer cells but also has a more intensified impact on normal cells causing increased and more severe toxicities [40]. A large population-based study showed that 62% of head and neck cancer patients receiving CRT and 46% of patients receiving RT alone had a hospitalization or ER visit for an acute adverse effect [1]. The most prominent side effect of RT or CRT is dysphagia. Dysphagia is the result of tissue fibrosis, mucositis, laryngopharyngeal dysmotility, and xerostomia [41]. Severe dysphagia can lead to malnutrition, aspiration, and pneumonia.

A complication unique to RT of the head and neck is ORN of the jaw. ORN of the jaw is a result of direct and indirect (loss of saliva) tissue effects that culminate in poor bone healing [42]. The patient can present with recurrent or chronic pain, mandible fracture, and exposed bone in the oral cavity. ORN of the jaw after RT is often precipitated by a dental procedure. Radiation-induced necrosis of cartilage is also a well-known complication of RT. For example, radiation-induced necrosis of the larynx can occur even years after treatment [43]. Differentiating radionecrosis of the larynx from recurrent cancer can be difficult. Diagnosis is based on examination and clinical suspicion.

Chemotherapy Complications

Patients receiving chemotherapy can manifest complications in the head and neck. For primary SCCHN, chemotherapy is often used in combination with RT for definitive treatment or alone a palliative therapy. Some common agents used for SCCHN include cisplatin, carboplatin, 5-fluorouracil, docetaxel, and cetuximab. These drugs can cause nausea and vomiting, renal failure, myelosuppression, thrombocytopenia, mucositis, and neuropathy. The majority of patients treated with CRT experience severe mucositis [44]. This can lead to decreased quality of life, weight loss, gastrostomy dependence, and increased ED visits and hospitalizations [45]. When symptoms are severe enough, up to one-third of SCCHN patients will require hospitalization [46]. The most common reason for presentation to the ED during or after CRT is dehydration and malnutrition. Symptoms can be ameliorated with topical lidocaine or "magic mouthwash," which usually includes topical lidocaine, steroid, antifungal,

and antibiotic. Finally, acute hearing loss can occur from the administration of chemotherapy, most notably cisplatin. Sudden-onset tinnitus may be an early sign of acute hearing loss. Steroids and discontinuation of cisplatin may limit the loss of hearing; however, cisplatin-induced hearing loss is generally permanent.

Common Pitfalls

Neck Abscess Versus Occult SCCHN Cystic Cervical Lymph Node Metastasis

The workup of a patient presenting to the ED with a neck mass often includes a contrast-enhanced CT of the neck. It is common that the only abnormality observed is a fluid-filled collection with peripheral rim enhancement (Fig. 19.8a, b). In an adult, this almost invariably represents occult SCCHN metastatic to a cystic or necrotic lymph node in the neck. It can unfortunately be misinterpreted as an abscess, particularly if there is redness of the skin and tenderness [47]. Unlike an abscess, these patients will usually lack the cardinal findings of infection, fever, and elevated WBC. Fine needle aspiration (FNA) biopsy is the most appropriate diagnostic test of any persistent neck mass >2 cm in an adult patient. Incision and drainage are strongly discouraged without a definitive diagnosis of abscess as it can substantially alter the management of a patient with occult SCCHN metastatic to a cervical lymph node.

Sinusitis Versus Occult Sinonasal Malignancy

Acute bacterial sinusitis is a common diagnosis for patients presenting to the ED. Sinonasal malignancies, on the other hand, are very rare. However, patients with cancer involving the sinuses are often treated unsuccessfully for sinusitis for weeks or months before an alternative diagnosis is entertained. The result is a delay in diagnosis that can impact the stage of disease, treatment options, and prognosis.

There are a few key differences between patients with sinusitis and sinonasal malignancy. First, the clinical presentation is strikingly different. Patients with sinonasal cancer most often present with unilateral, rather than bilateral symptoms. Unilateral nasal obstruction, persistent nasal bleeding, facial pain or pressure, facial numbness, visual changes, and/ or epiphora should be carefully evaluated for a possible sinonasal malignancy. This could most readily be accomplished with a sinus CT. Any unilateral opacification of the sinuses should prompt timely referral to an otolaryngologist (Fig. 19.9).

Ear Infection Versus Occult SCCHN of the Oropharynx

Unilateral otalgia is a common presenting symptom of SCCHN involving the oropharynx, including the tonsil or base of tongue. Other common causes of ear pain are infection and temporal mandibular joint (TMJ) disorder. It is



Fig. 19.8 Patient presenting with a non-tender left-sided neck mass (a) which appears cystic on axial computed tomography (CT) imaging (b). Any lateral neck mass in an adult should be considered cancer until proven otherwise, regardless of smoking status or other risk factors



Fig. 19.9 Coronal computed tomography (CT) of the sinuses demonstrating a left-sided nasal cavity mass with opacification of the left maxillary sinus and ethmoid sinuses and invasion into the ipsilateral orbit. Any unilateral opacification of the sinuses is concerning for malignancy

important for medical practitioners in the ED to be able to distinguish between benign and malignant causes of ear pain. As with occult sinonasal malignancies, a delay in diagnosis is common for patients with SCCHN of the oropharynx.

The incidence of SCCHN involving the oropharynx (tonsil or base of tongue) continues to increase dramatically [48]. This has been attributed to the human papillomavirus (HPV) which is now associated with >70% of these cases. Patients with HPV-associated SCCHN tend to be younger and are more often non-smokers and are often not thought to be at risk of head and neck cancer [49]. However, unilateral otalgia without clinical findings of an ear infection (e.g., ear drainage, middle ear effusion, painful ear canal) should prompt a thorough evaluation of the oropharynx by a specialist to rule out occult malignancy.

Summary

The UADT is a complex area where functions of breathing and eating take place in a highly vascularized area. Patients with SCCHN frequently present to the ED with a difficult airway. Frontline physicians in the ED must have a plan for airway control in these patients. An understanding of anatomy and physiology of the surgically altered airway in patients after tracheostomy and laryngectomy, so-called neck breathers, is essential. Bleeding related to the treatment of SCCHN can be catastrophic. Therefore, control of the airway along with methods to stabilize the patient is important. The use of direct pressure to tamponade bleeding and shock trauma principles can be employed before definitive management by a head and neck surgeon. There are other treatment-related complications related to surgery, radiation therapy, and chemotherapy that must be recognized to prevent further sequelae. To avoid common pitfalls in SCCHN management, ED medical personnel must also be aware of distinctions between signs of malignancy and common otolaryngologic symptoms.

References

- O'Neill CB, Baxi SS, Atoria CL, O'Neill JP, Henman MC, Sherman EJ, et al. Treatment-related toxicities in older adults with head and neck cancer: a population-based analysis. Cancer. 2015;121(12):2083–9.
- Zevallos JP. International head and neck cancer epidemiology consortium: update no. 18. Head Neck. 2015;37(7):931–2.
- Eskander A, Krzyzanowska MK, Fischer HD, Liu N, Austin PC, Irish JC, et al. Emergency department visits and unplanned hospitalizations in the treatment period for head and neck cancer patients treated with curative intent: a population-based analysis. Oral Oncol. 2018;83:107–14.
- Tang PL, Cheng JS, Huang WC, Chang HS, Chen HC. Why do head and neck cancer patients visit the emergency department? Am J Emerg Med. 2015;33(8):1102–5.
- Reyes-Gibby CC, Melkonian SC, Hanna EY, Yeung SJ, Lu C, Chambers MS, et al. Cohort study of oncologic emergencies in patients with head and neck cancer. Head Neck. 2017;39(6):1195–204.
- Varghese BT, Balakrishnan M, Kuriakose R. Fibre-optic intubation in oncological head and neck emergencies. J Laryngol Otol. 2005;119(8):634–8.
- Chen K, Varon J, Wenker OC. Malignant airway obstruction: recognition and management. J Emerg Med. 1998;16(1):83–92.
- Shen HN, Cheng KC, Hou CC, Tan CK, Huang WT. Clinical features and short-term outcome of critically ill patients with head and neck cancer in the medical intensive care unit. Am J Clin Oncol. 2009;32(5):467–71.
- Xu B, Boero IJ, Hwang L, Le QT, Moiseenko V, Sanghvi PR, et al. Aspiration pneumonia after concurrent chemoradiotherapy for head and neck cancer. 2015;121(8):1303–11.
- Caranci F, Brunese L, Reginelli A, Napoli M, Fonio P, Briganti F. Neck neoplastic conditions in the emergency setting: role of multidetector computed tomography. Semin Ultrasound CT MR. 2012;33(5):443–8.
- McGarvey JM, Pollack CV. Heliox in airway management. Emerg Med Clin North Am. 2008;26(4):905–20, viii.
- Chin BS, Sabin S, Erickson A, Ghosh BC. Emergency tracheostomy for advanced head and neck tumor. J Surg Oncol. 1998;67(1):49–51.
- Standley TD, Smith HL. Emergency tracheal catheterization for jet ventilation: a role for the ENT surgeon? J Laryngol Otol. 2005;119(3):235–6.
- 14. Smallridge RC, Ain KB, Asa SL, Bible KC, Brierley JD, Burman KD, American Thyroid Association Anaplastic Thyroid Cancer Guidelines Taskforce, et al. American Thyroid Association guidelines for management of patients with anaplastic thyroid cancer. Thyroid. 2012;22(11):1104–39.
- Mani N, McNamara K, Lowe N, Loughran S, Yap BK. Management of the compromised airway and role of tracheotomy in anaplastic thyroid carcinoma. Head Neck. 2016;38(1):85–8.

- Arola MK. Tracheostomy and its complications. A retrospective study of 794 tracheostomized patients. Ann Chir Gynaecol. 1981;70(3):96–106.
- Al-Samri M, Mitchell I, Drummond DS, Bjornson C. Tracheostomy in children: a population-based experience over 17 years. Pediatr Pulmonol. 2010;45(5):487–93.
- Townsley RB, Baring DE, Clark LJ. Emergency department care of a patient after a total laryngectomy. Eur J Emerg Med. 2014;21(3):164–9.
- Graboyes EM, Yang Z, Kallogjeri D, Diaz JA, Nussenbaum B. Patients undergoing total laryngectomy: an at-risk population for 30-day unplanned readmission. JAMA Otolaryngol Head Neck Surg. 2014;140(12):1157–65.
- Leuin SC, Deschler DG. The missing tracheoesophageal puncture prosthesis: evaluation and management. Ear Nose Throat J. 2013;92(2):E14–6.
- Liang W, Guan W, Chen R, Wang W, Li J, Xu K, et al. Cancer patients in SARS-CoV-2 infection: a nationwide analysis in China. Lancet Oncol. 2020;21(3):335–7.
- American Academy of Otolaryngology-Head Neck Surgery. AAO-HNS COVID-19 Resources. Guidance for return to practice for Otolaryngology-Head and Neck Surgery. 8 May 2020. [Internet]. https://www.entnet.org/content/guidance-returnpractice-otolaryngology-head-and-neck-surgery. Accessed 15 Oct 2020.
- Cheung JC, Ho LT, Cheng JV, Cham EY, Lam KN. Staff safety during emergency airway management for COVID-19 in Hong Kong. Lancet Respir Med. 2020;8(4):e19.
- Crossley J, Clark C, Brody F, Maxwell JH. Surgical considerations for an awake tracheotomy during the COVID-19 pandemic. J Laparoendosc Adv Surg Tech A. 2020;30(5):477–80.
- 25. Miles BA, Schiff B, Ganly I, Ow T, Cohen E, Genden E, et al. Tracheostomy during SARS-CoV-2 pandemic: recommendations from the New York Head and Neck Society. Head Neck. 2020;42(6):1282–90.
- 26. Roh JL, Suh DC, Kim MR, Lee JH, Choi JW, Choi SH, et al. Endovascular management of carotid blowout syndrome in patients with head and neck cancers. Oral Oncol. 2008;44(9):844–50.
- Kortbeek JB, Al Turki SA, Ali J, Antoine JA, Bouillon B, Brasel K, et al. Advanced trauma life support, 8th edition, the evidence for change. J Trauma. 2008;64(6):1638–50.
- Okamura HO, Kamiyama R, Takiguchi Y, Kimizuka K, Ishikawa N, Kishimoto S. Histopathological examination of ruptured carotid artery after irradiation. ORL J Otorhinolaryngol Relat Spec. 2002;64(3):226–8.
- Mazumdar A, Derdeyn CP, Holloway W, Moran CJ, Cross DT 3rd. Update on endovascular management of the carotid blowout syndrome. Neuroimaging Clin N Am. 2009;19(2):271–81.
- Kakizawa H, Toyota N, Naito A, Ito K. Endovascular therapy for management of oral hemorrhage in malignant head and neck tumors. Cardiovasc Intervent Radiol. 2005;28(6):722–9.
- Risley J, Mann K, Jones NS. The role of embolisation in ENT: an update. J Laryngol Otol. 2012;126(3):228–35.
- Timon CV, Brown D, Gullane P. Internal jugular vein blowout complicating head and neck surgery. J Laryngol Otol. 1994;108(5):423–5.
- Takasaki K, Enatsu K, Nakayama M, Uchida T, Takahashi H. A case with tracheo-innominate artery fistula. Successful manage-

ment of endovascular embolization of innominate artery. Auris Nasus Larynx. 2005;32(2):195-8.

- Epstein SK. Late complications of tracheostomy. Respir Care. 2005;50(4):542–9.
- D'Aguanno V, Ralli M, Greco A, de Vincentiis M. Clinical recommendations for epistaxis management during the COVID-19 pandemic. Otolaryngol Head Neck Surg. 2020;163(1):75–7.
- Smith BG, Hutcheson KA, Little LG, Skoracki RJ, Rosenthal DI, Lai SY, Lewin JS. Lymphedema outcomes in patients with head and neck cancer. Otolaryngol Head Neck Surg. 2015;152(2):284–91.
- 37. Roediger FC, Eisele DW. Complications of neck surgery. In: Flint PW, Haughey BH, Lund VJ, Niparko JK, Richardson MA, Robbins KT, Thomas JR, editors. Cummings otolaryngology, head & neck surgery. 5th ed. Philadelphia: Mosby Elsevier; 2010. p. 1726–34.
- Sayles M, Grant DG. Preventing pharyngo-cutaneous fistula in total laryngectomy: a systematic review and meta-analysis. Laryngoscope. 2014;124(5):1150–63.
- Allen CT, Lee CJ, Merati AL. Clinical assessment and treatment of the dysfunctional larynx after radiation. Otolaryngol Head Neck Surg. 2013;149(6):830–9.
- 40. Jensen SB, Pedersen AM, Vissink A, Andersen E, Brown CG, Davies AN, Salivary Gland Hypofunction/Xerostomia Section, Oral Care Study Group, Multinational Association of Supportive Care in Cancer (MASCC)/International Society of Oral Oncology (ISOO), et al. A systematic review of salivary gland hypofunction and xerostomia induced by cancer therapies: prevalence, severity and impact on quality of life. Support Care Cancer. 2010;18(8):1039–60.
- Dworkin JP, Hill SL, Stachler RJ, Meleca RJ, Kewson D. Swallowing function outcomes following nonsurgical therapy for advancedstage laryngeal carcinoma. Dysphagia. 2006;21(1):66–74.
- Kim MG, Lee ST, Park JY, Choi SW. Reconstruction with fibular osteocutaneous free flap in patients with mandibular osteoradionecrosis. Maxillofac Plast Reconstr Surg. 2015;37(1):7.
- Fitzgerald PJ, Koch RJ. Delayed radionecrosis of the larynx. Am J Otolaryngol. 1999;20(4):245–9.
- 44. Lewis SL, Brody R, Touger-Decker R, Parrott JS, Epstein J. Feeding tube use in patients with head and neck cancer. Head Neck. 2014;36(12):1789–95.
- 45. Allison RR, Ambrad AA, Arshoun Y, Carmel RJ, Ciuba DF, Feldman E, et al. Multi-institutional, randomized, double-blind, placebo-controlled trial to assess the efficacy of a mucoadhesive hydrogel (MuGard) in mitigating oral mucositis symptoms in patients being treated with chemoradiation therapy for cancers of the head and neck. Cancer. 2014;120(9):1433–40.
- 46. Murphy BA, Beaumont JL, Isitt J, Garden AS, Gwede CK, Trotti AM, et al. Mucositis-related morbidity and resource utilization in head and neck cancer patients receiving radiation therapy with or without chemotherapy. J Pain Symptom Manag. 2009;38(4):522–32.
- Chuang SY, Lin HT, Wen YS, Hsu FJ. Pitfalls of CT for deep neck abscess imaging assessment: a retrospective review of 162 cases. B-ENT. 2013;9(1):45–52.
- Chaturvedi AK, Engels EA, Pfeiffer RM, Hernandez BY, Xiao W, Kim E, et al. Human papillomavirus and rising oropharyngeal cancer incidence in the United States. J Clin Oncol. 2011;29(32):4294–301.
- 49. D'Souza G, Kreimer AR, Viscidi R, Pawlita M, Fakhry C, Koch WM, et al. Case-control study of human papillomavirus and oropharyngeal cancer. N Engl J Med. 2007;356(19):1944–56.



20

Ophthalmic Emergencies in Cancer Patients

Diana Chao, Mathieu F. Bakhoum, and Bita Esmaeli

Introduction

While there are a few ophthalmological emergencies that are life-threatening, there are a number of ocular conditions in cancer patients that require immediate diagnosis and management. This chapter reviews the differential diagnosis and management of common ocular or visual symptoms encountered in a cancer hospital-based emergency department (ED). It is organized based on the symptoms that patients may present with to the ED, including acute visual loss, diplopia, red eye, proptosis, epiphora, ptosis, flashes, and floaters. A review of trauma-related ocular emergencies is outside the scope of this textbook, and the reader is referred to other texts for a detailed discussion of noncancer-related ocular emergencies [1, 2].

Acute Visual Loss

One of the most distressing ophthalmologic symptoms is the sudden loss of vision. The causes of sudden acute visual loss may be classified as those affecting the optic nerve, those affecting the retina, and those affecting the retinal vasculature.

D. Chao Intelligent Eye Care, Arcadia, CA, USA

Ophthalmic Plastic and Orbital Surgery, Department of Plastic Surgery, MD Anderson Cancer Center, Houston, TX, USA

M. F. Bakhoum

Department of Ophthalmology, University of California, San Diego, La Jolla, CA, USA

Ophthalmic Plastic and Orbital Surgery, Department of Plastic Surgery, MD Anderson Cancer Center, Houston, TX, USA

B. Esmaeli (🖂)

Ophthalmic Plastic and Orbital Surgery, Department of Plastic Surgery, MD Anderson Cancer Center, Houston, TX, USA e-mail: besmaeli@mdanderson.org The primary symptoms associated with optic nerve disease may include decreased visual acuity associated with a central visual field defect, decreased color vision and contrast sensitivity, and ocular pain on eye movement. A sensitive clinical sign for the presence of asymmetric optic nerve disease is a relative afferent pupillary defect (Marcus Gunn pupil) on the affected side using the swinging-light test [3]. On ophthalmoscopy, the optic nerve head may appear swollen or pale. A red desaturation test may also be performed; when there is damage to the optic nerve, the affected eye sees the color red as a blanched orange-pink color.

Acute visual loss secondary to optic nerve disease in cancer patients can be due to a mass effect, either from a primary orbital tumor process (Fig. 20.1), secondary to an orbital metastatic process (Fig. 20.2), or from secondary extension



Fig. 20.1 Axial T2 image demonstrates an orbital lymphoma compressing the optic nerve and leading to visual loss. (From Debnam [73], with permission Springer Nature)



Fig. 20.2 Axial T2 image demonstrates metastasis from clear cell renal cell carcinoma to the right lateral rectus muscle. The patient presented with right eye pain, proptosis, diplopia, and blurry vision



Fig. 20.3 Computed tomography sagittal plane demonstrates a nasopharyngeal carcinoma invading the orbit posteriorly through the inferior orbital fissure. (From Shinder and Esmaeli [74], with permission Springer Nature)

of tumor from paranasal sinuses (Fig. 20.3), from nasal cavity, or from the brain or skull base. Primary malignancies of the optic nerve include optic nerve glioma (Fig. 20.4), meningioma, craniopharyngioma, and medulloblastoma [4].



Fig. 20.4 Axial T2 image demonstrates left optic nerve glioma in a 61-year-old female who presented with visual loss in the left eye



Fig. 20.5 Fundus photograph demonstrates infiltration of the optic nerve by leukemic cells, causing progressive visual loss

Infiltration of the optic nerve by leukemic or lymphomatous cells may also occur (Fig. 20.5) [5]. In addition, the optic nerve may be infiltrated by leptomeningeal disease from solid or liquid tumors. Invasive aspergillosis of the paranasal sinuses and/or orbit should also be considered in immuno-compromised patients, as it is most prevalent among leukemic patients with granulocytopenia and is associated with a high mortality rate [6]. Optic nerve toxicity secondary to chemotherapeutic agents is another possible cause of optic neuropathy in cancer patients.

The most common noncancer-related cause of optic neuritis (optic nerve swelling) is multiple sclerosis; however, optic neuritis may also result from inflammatory conditions,



Fig. 20.6 Fundus photograph demonstrates ischemic optic neuropathy in a patient with diabetes mellitus who experienced acute painless loss of vision

such as Wegener's granulomatosis (granulomatosis with polyangiitis), systemic lupus erythematosus, and sarcoidosis, or it may be idiopathic. Infectious etiologies (including syphilis and Lyme disease) may also produce similar findings. In the elderly population or in patients with atherosclerosis, hypertension, and/or diabetes, the most common cause of acute visual loss of optic nerve origin is an ischemic optic neuropathy (Fig. 20.6). In the older population, giant cell arteritis is an important form of ischemic optic neuropathy that is sometimes associated with polymyalgia rheumatica [7]. Giant cell arteritis requires prompt diagnosis and treatment with high-dose systemic steroids to prevent progressive and sometimes bilateral visual loss. In diabetic patients or in the immunocompromised cancer patients, the possibility of orbital cellulitis or fungal infections such as mucormycosis or aspergillosis of the sinus with extension into the orbit should also be considered as a cause of optic nerve swelling and compression.

The preferred diagnostic test to evaluate optic nerve disease is magnetic resonance imaging (MRI) of the brain and orbit, with and without gadolinium, fat suppressed. MRI can usually demonstrate the extent of optic nerve disease, although may be normal in the early stages of leptomeningeal disease [8].

The management of optic nerve disease depends on the etiology. In cancer patients, the initial management may include systemic antibiotics or antifungals, chemotherapy, external beam radiation therapy, high-dose steroids, or surgery [9-11].

Retinal disease (particularly if it involves the macula, where visual acuity is most sensitive) may cause acute visual loss. Symptoms associated with retinal disease include decreased vision, metamorphopsia, flashes of light, new floaters, and a "curtain" over the visual field. A dilated fundus examination is necessary to correctly diagnose the retinal causes of acute visual loss.

Rhegmatogenous retinal detachment is the most common type of detachment, and it occurs when there is a tear or break in the retina, allowing fluid to accumulate in the subretinal space, separating the neurosensory retina from the underlying retinal pigment epithelium. In cancer patients, serous and exudative retinal detachments may result from leukemic or lymphomatous infiltration of the choroid and/or subretinal space, choroidal metastatic lesions (Fig. 20.7), or, less commonly, primary intraocular tumors such as uveal melanoma. Serous and exudative detachments occur despite the absence of a hole, tear, or break. Opportunistic infections such as those with *Cytomegalovirus* (CMV) (Fig. 20.8), herpes simplex virus (HSV), herpes zoster virus (HZV), and *Candida* may cause retinitis in immunocompromised patients. Retinitis due to HSV or HZV may cause rapid



Fig. 20.7 Fundus photograph demonstrates a choroidal metastatic lesion causing elevation of the choroid and metamorphopsia



Fig. 20.8 Fundus photograph demonstrates CMV retinitis, characterized by necrosis and hemorrhage often in the posterior pole



Fig. 20.9 Fundus photograph demonstrates spontaneous retinal hemorrhage in a patient with acute myeloid leukemia and thrombo-cytopenia

visual loss to "no light perception" within 24 h [12]. It is important to diagnose the infectious forms of retinitis in a timely fashion so that the appropriate systemic therapy can be initiated as soon as possible.

Another very common cause of visual loss among cancer patients is retinal hemorrhage secondary to thrombocytopenia (Fig. 20.9) [13]. Prompt referral to an ophthalmologist is necessary for appropriate diagnosis and management. Treatment for rhegmatogenous detachments includes urgent laser photocoagulation or surgical management. For all other etiologies, timely treatment of underlying disease including chemotherapy and/or radiation for metastatic lesions and intravenous and/or intravitreal antiviral and antifungal therapy for opportunistic viral infections is necessary.

Obstruction of the retinal vasculature can also cause acute visual loss. Retinal vascular obstruction usually results from thrombi or emboli and is more likely to occur in patients with hypertension, atherosclerosis, or diabetes. However, cancer patients have the added risk of neoplasm-associated hypercoagulability [14]. Central retinal artery occlusion (CRAO) or central retinal vein occlusion (CRVO) may lead to devastating visual loss. Involvement of the smaller vessels may lead to partial visual acuity or visual field loss. Most occlusive vascular events are not reversible although they require prompt diagnosis and follow-up to address the underlying medical problems and to prevent future ocular complications from ischemic retinopathy. The visual prognosis depends on the extent of retinal ischemia. Particularly, the ischemic variety of CRVO can be complicated by secondary neovascular glaucoma and may require panretinal photocoagulation and intravitreal anti-VEGF inhibitors [15]. Prompt referral to an ophthalmologist is recommended when a retinal vascular event is suspected.

Diplopia

Diplopia (double vision) is a common symptom in cancer patients. The first thing to establish is whether diplopia is monocular or binocular and whether it is horizontal or vertical. In addition, true diplopia must be distinguished from blurred vision, in which the image is blurred, but is not in fact double. If diplopia persists after one eye is covered, the patient has monocular diplopia, which is almost certainly not a neurologic problem. The usual causes of monocular diplopia include refractive error or media opacity (i.e., cataract). In contrast, if diplopia is present only when both eyes are open, it is binocular in nature, and there is usually an underlying neurologic or extraocular motility problem. Binocular diplopia can be horizontal, vertical, or torsional [16]. Extraocular motility exam should be performed to assess whether only one or multiple cranial nerves are affected and for any signs of ptosis.

Specific neurologic causes of binocular diplopia include cranial nerve III, IV, or VI palsies, or a mechanical process that may limit the function of the extraocular muscles. If only one cranial nerve or extraocular muscle is affected, then a simple noncomitant diplopia may develop. In contrast, leptomeningeal disease or any space-occupying lesion in the orbital apex, superior orbital fissure, or the cavernous sinus may affect multiple cranial nerves at the same time, resulting in a more complex pattern of diplopia.

While cranial nerve palsies can be secondary to an ischemic event (particularly in patients with hypertension, diabetes, or atherosclerosis), in cancer patients, the most common etiology is tumor extension in the orbital apex or cavernous sinus. Extraocular muscles may also be compressed or entrapped by a mass, or they may be infiltrated by inflammatory or neoplastic processes. Opportunistic infections, particularly fungal infections secondary to mucormycosis or aspergillosis, may extend into the orbit from the paranasal sinuses. A high index of suspicion for fungal cellulitis or pansinusitis is necessary to make the correct diagnosis and initiate therapy for these potentially fatal infections in immunocompromised patients [17].

Another important but less common cause of third cranial nerve palsy, particularly if pupillary fibers are involved, is a cerebral aneurysm. This is not unique to cancer patients but should be considered on the differential diagnosis of a patient with an acute onset of third cranial nerve palsy.

In the emergency evaluation of a patient with an acute onset of diplopia, an imaging study (ideally, an MRI of the brain and orbit with and without gadolinium, fat suppressed) is often necessary to rule out or establish the diagnosis and extent of orbital or cavernous sinus involvement if any.

If a cerebral aneurysm is suspected, magnetic resonance angiogram (MRA) or computed tomography angiography (CTA) of the brain is indicated. The treatment of diplopia depends on the underlying cause and (usually in cancer patients) entails treatment of the underlying tumor or infectious etiology. The patching of one eye or temporary Fresnel prisms may help the patient symptomatically until the exact cause and treatment for diplopia is determined.

Red Eye

There are many possible causes of a red eye in cancer patients. It is helpful to classify the causes of a red eye on the basis of intraocular structures that may be the cause of inflammation on the surface of the globe. Any disease process that can cause inflammation in the conjunctiva, cornea, iris, anterior chamber, ciliary body, or sclera can present as a red eye. Therefore, it is important to perform a comprehensive ophthalmologic examination to identify the correct cause.

Conjunctivitis is probably the most common cause of a red eye. Conjunctivitis can be due to infectious etiology such as bacteria (*Staphylococcus aureus*, *Streptococcus pneumoniae*, *Pseudomonas aeruginosa*, *Chlamydia*, *Neisseria gonococcus*) or viruses (adenovirus, herpes simplex virus [HSV]) [18]. However, occasionally, the cause is not infectious. The inflammatory causes of conjunctivitis include toxic conjunctivitis secondary to the excessive or inappropriate use of topical antibiotics (medicamentosa), allergic conjunctivitis, and acute or chronic ocular graft-versus-host disease.

Subconjunctival hemorrhage and hemorrhagic chemosis are benign conditions that may cause asymptomatic, sudden, painless red eye (Fig. 20.10). Conservative management



Fig. 20.10 Subconjunctival hemorrhage, causing painless red eye, in a 57-year-old patient who has a history of acute lymphoblastic leukemia and low platelet count $(30,000/\mu L)$

with lubrication of the eye and observation for spontaneous resolution over several weeks is appropriate.

Superficial keratopathy secondary to ocular graft-versushost disease (GVHD) or as a side effect of cancer chemotherapeutic agents such as arabinosylcytosine (ara-C) can cause ocular surface irritation, corneal epithelial defects, and, possibly, a red eye [19, 20]. Many other chemotherapeutic agents such as docetaxel and 5-fluorouracil are secreted in the tear film and may lead to conjunctivitis. Slit-lamp examination of the cornea using topical fluorescein dye and cobaltblue light to assess for decreased tear film, punctate epithelial erosions, pseudomembranes, serosanguinous exudates, and corneal epithelial defects is performed to evaluate for most forms of superficial keratopathy. The management of conjunctival or corneal problems secondary to ocular GVHD entails the use of topical lubricants, cyclosporine drops, punctal plugs, and topical and/or systemic administration of immunosuppressive agents, such as steroids and tacrolimus [21]. Superficial keratopathy secondary to ara-C use is treated with topical steroids, lubricating artificial tears, and ophthalmic ointment, and in refractory cases by lowering the dose of ara-C.

Another common cause of superficial keratopathy in cancer patients is exposure keratopathy secondary to facial palsy. Facial paralysis secondary to the compressive effects of a parotid mass or due to ablative surgery for malignancies in the parotid area can result in inadequate eyelid closure and chronic corneal exposure [22]. The immediate treatment of exposure keratopathy entails the use of lubricating artificial tears and ophthalmic ointments. If facial paralysis is expected to last longer than a few weeks, periocular surgery, such as placement of a gold weight in the upper eyelid, repair of paralytic lower eyelid, and a lateral tarsorrhaphy, should be considered [23].

Infectious keratitis is another important cause of red eye and can be caused by bacterial (S. aureus, S. pneumoniae, N. gonococcus, Moraxella, P. aeruginosa), viral (adenovirus, HSV, HZV), or fungal (Candida, Aspergillus) organisms [24]. Herpes zoster ophthalmicus occurs when varicella-zoster virus is reactivated in the ophthalmic (V1) division of the trigeminal nerve and is associated with immunosuppression, and it may be a harbinger of increased risk of cancer [25] or other immunocompromised states. It may present with periorbital vesicle formation in a unilateral, V1 dermatomal distribution with associated red eye and pseudodendritic fluorescein-staining pattern under cobalt light. Prompt diagnosis and treatment of HZV with systemic antiviral agents like acyclovir and topical agents is appropriate. The diagnosis and management of infectious keratitis requires the direct involvement of an ophthalmologist. Management often involves obtaining corneal cultures and instituting topical fortified antibacterial, antiviral, or antifungal agents.

Acute angle-closure glaucoma may also cause a painful red eye. It occurs in patients who have narrow angles that become blocked by the iris. Symptoms and signs of acute angle-closure glaucoma are pain, redness, blurred, or "steamy" vision from corneal edema, halos around lights, and a mid-dilated pupil. The intraocular pressure can rise to 50-60 mmHg, and urgent medical treatment to lower the pressure is necessary to avoid permanent vision loss. There are many causes of acute angle-closure glaucoma, and prompt diagnosis is necessary for appropriate treatment. Anticholinergic or sympathomimetic medications dilate the iris and lead to crowding of the anterior chamber angle peripherally [26]. A mass in the ciliary body or choroid can also push the iris forward and cause angle closure [27]. Primary intraocular tumors such as uveal melanomas or medulloepitheliomas may be present in the angle and may obstruct the aqueous outflow. The medical management of angle-closure glaucoma in the ED includes the immediate use of topical antiglaucoma medications, and systemic carbonic anhydrase inhibitors are often necessary to bring the intraocular pressure down to a safe level. For primary acute angle-closure glaucoma, the patient should be referred to an ophthalmologist for consideration of a laser peripheral iridotomy after the intraocular pressure has been brought down to a safer level with medications [28].

Inflammation of the iris, ciliary body, or choroid is referred to as uveitis. In addition to red eye, uveitis can present with pain, photophobia, blurred vision, and miosis. Anterior chamber cells and flare noted during slit-lamp biomicroscopy are the hallmarks of iritis or uveitis. Uveitis is thought to be idiopathic in about 50% of cases or can be associated with various autoimmune processes such as rheumatoid arthritis, lupus, ankylosing spondylitis, and Wegener's granulomatosis (granulomatous polyangiitis) [29]. In immunocompromised patients, infectious causes of uveitis must be considered. The most severe form of intraocular infection, endogenous endophthalmitis, can initially present as mild but progressively worsening uveitis [30]. Once the diagnosis of endogenous endophthalmitis is suspected, blood culture and vitreous biopsy are often necessary to identify the causative infectious organism. Prompt referral to an ophthalmologist is necessary for diagnosis and management. Treatment for uveitis due to noninfectious causes includes the judicious use of topical steroids and cycloplegic drops to decrease the inflammation in the anterior chamber and prevent ciliary body spasm. For suspected endogenous endophthalmitis, broad-spectrum intravenous antimicrobial therapy is administered until sensitivity results are available from the vitreous biopsy cultures. Intravitreal injection of antibiotics or antifungal drugs is the treatment for endogenous endophthalmitis, and in some cases, a surgical vitrectomy may be necessary both for diagnostic and for therapeutic purposes [31, 32].

Epiphora

True epiphora (excessive tearing) results from an obstruction of the tear drainage apparatus. Epiphora must be differentiated from pseudoepiphora, which may be caused by ocular surface irritation due to conditions such as dry eye syndrome, ocular graft-versus-host disease, and exposure keratopathy. The most common cause of epiphora in the general population is primary nasolacrimal duct blockage which is due to obstruction of the nasolacrimal duct at its junction with the lacrimal sac [33]. Primary idiopathic nasolacrimal duct blockage occurs more frequently in women and is involutional in nature. In cancer patients, however, the most common etiology for epiphora is likely to be (a) mechanical blockage of the tear drainage pathway secondary to either primary lacrimal sac or nasolacrimal duct cancers or from extension of tumors from the paranasal sinus or nasal cavity [34], (b) canalicular and nasolacrimal duct stenosis secondary to chemotherapy [35, 36], or (c) canalicular or nasolacrimal duct obstruction secondary to radiation therapy [37, 38]. Common chemotherapeutic agents that are known to cause canalicular stenosis include S-1, 5-fluorouracil, mitomycin C, and docetaxel (Taxotere) [39-42]. Nasolacrimal duct obstruction may also occur secondary to local toxicity or active uptake of radioactive iodine used in the treatment of thyroid carcinoma [43-45]. Because timely diagnosis of early canalicular stenosis in patients receiving these agents can lead to early insertion of silicone tubing in the nasolacrimal duct and therefore prevention of further narrowing of the canaliculi, appropriate referral to an ophthalmologist early in the course of therapy with these agents is crucial.

Acute dacryocystitis is another important cause of epiphora. The infectious causes for acute or chronic dacryocystitis include *S. aureus*, *S. pneumoniae*, and *Haemophilus influenzae* [46]. Clinical signs associated with dacryocystitis are epiphora, mucopurulent discharge upon pressure over the lacrimal sac, and erythema and edema over the lacrimal sac. Initial treatment for infectious dacryocystitis involves systemic antibiotics and warm compresses. A dacryocystorhinostomy may be necessary to prevent future episodes of dacryocystitis, particularly in immunocompromised patients [47].

Proptosis

Proptosis (outward protrusion of the eye) may be caused by an orbital mass or a diffuse inflammatory or infiltrative process involving the retrobulbar space. Other possible associated symptoms and signs may include diplopia, decreased vision, and multiple cranial neuropathies secondary to involvement of the orbital apex.



Fig. 20.11 Axial T1 image demonstrates a right lacrimal gland lymphoma in a 75-year-old female who presented with a slow-growing painless mass and mild proptosis

The most common primary cancer affecting the orbit in adults is lymphoma (Fig. 20.11) [48, 49]. Orbital lymphoma can be the extranodal manifestation of systemic lymphoma or may be the only site of lymphomatous involvement [50, 51]. Other benign or malignant tumors that can cause proptosis include optic nerve glioma, meningioma, orbital hemangioma, sarcoma, and metastatic lesions [52]. The most important cause of sudden and progressive proptosis in children is orbital rhabdomyosarcoma [53]. Another important cause of proptosis, particularly if associated with pain and inflammatory signs, is orbital pseudotumor. The diagnosis of orbital inflammatory syndrome (orbital pseudotumor) is a diagnosis of exclusion and should be made only after an orbital biopsy specimen proves to be negative for malignancy [54]. Timely diagnosis with diagnostic imaging (preferably MRI of the brain and orbit with and without gadolinium, fat suppressed) and orbital biopsy is important for institution of appropriate therapy. Management of proptosis consists of treatment of the underlying cause. It is important to avoid the administration of systemic steroids until the diagnosis is clearly established, ideally on the basis of an orbital biopsy in addition to MRI to rule out lymphoma, orbital metastasis, or rhabdomyosarcoma as the underlying cause of proptosis. The use of anti-inflammatory agents can mask the clinical signs and symptoms, delay diagnosis, and lead to a lower yield for an orbital biopsy. Prompt referral to an orbital and oculoplastic surgeon is appropriate when a patient presents with acute proptosis. An experienced orbital specialist may be able to recognize common radiographic features of common orbital lesions and sometime avoid a biopsy, but in most instances, if radiographic features are not classic for a benign vascular lesion such as hemangioma, the most appropriate next step after an imaging study is an orbital biopsy or complete excision of the mass depending on the radiographic features.

Orbital cellulitis may also present as proptosis, and it is associated with visual loss, decreased and painful extraocular movements, and general orbital congestion. Orbital cellulitis usually results from the direct extension of infection from the paranasal sinuses, especially the ethmoidal sinus [55]. However, direct inoculation from trauma, extension of an eyelid infection, and septicemia may also cause orbital cellulitis [56]. The causative infectious organisms are typically S. aureus, H. influenzae, S. pneumoniae, or fungi, such as Aspergillus [57]. Orbital cellulitis may be complicated by the formation of an orbital abscess or by direct extension of infection into the cavernous sinus and the brain, a complication with a high risk of mortality [58]. Immediate treatment with systemic antibiotics and antifungals is prudent when orbital cellulitis or an orbital abscess is suspected. An orbital abscess can be diagnosed by computed tomography (CT) or MRI and usually requires immediate surgical drainage, particularly if it is associated with progressive visual loss or proptosis [59].

Proptosis can also be caused by orbital hemorrhage. Possible causes include postoperative hemorrhage, trauma, and hematologic disorders [60]. The patient's vision and intraocular pressure (IOP) should be immediately assessed because retrobulbar hemorrhage may cause a compressive optic neuropathy that may lead to permanent visual loss. If the vision is decreased or the IOP is elevated above 30-35 mmHg, a lateral canthotomy and cantholysis should be considered in the ED to expand the orbital volume and relieve orbital pressure [61]. In the pancytopenic cancer patients who are often pancytopenic due to chemotherapy, more conservative measures such as the use of pressurelowering glaucoma drops may be more appropriate and should be tried first as a canthotomy may lead to continuous oozing and bleeding from the orbit. Orbital emphysema can rarely yield findings similar to those of orbital hemorrhage, and a lateral canthotomy and cantholysis or needle decompression may also be indicated if symptoms and signs of compartment syndrome of the orbit occur [62, 63]. The usual cause of orbital emphysema is trauma or a history of tracheal or thoracic surgery [64].

Ptosis

Ptosis (droopiness of the upper eyelid) can be gradual or sudden in onset. As with other symptoms discussed in this chapter, determining the underlying cause is the most important aspect of the management of ptosis in the ED. The most common cause of ptosis in adults in the general population is involutional ptosis. In children, a congenital abnormality of the levator muscle is the most common cause of ptosis. In cancer patients, the most common cause of ptosis is neurologic. A palsy of the third cranial nerve due to primary or metastatic tumors of the base of the skull can cause ptosis, decreased extraocular muscle movement, and a dilated pupil. Perineural invasion secondary to cutaneous carcinomas of the facial skin can also cause multiple cranial neuropathies,

including a third nerve palsy [65].

Another neurologic cause of ptosis is Horner syndrome. Horner syndrome refers to the triad of ipsilateral mild ptosis $(\leq 2 \text{ mm})$, miosis of the pupil, and anhidrosis [66]. A mass effect anywhere along the path of sympathetic fibers can cause Horner syndrome. This three-neuron chain originates in the hypothalamus. The second-order neurons originate in Budge's center (C8-T2) and wind over the lung apex. The third-order neurons originate in the superior cervical ganglion, where they follow the carotid artery and then the fifth and sixth cranial nerves before they accompany the third cranial nerve to the eye. When ipsilateral miosis is associated with ptosis, Horner syndrome must be ruled out. Associated signs and symptoms may be helpful in determining the location of the lesion causing Horner syndrome. For example, ataxia, nystagmus, and weakness may indicate a first-order Horner syndrome from a brain tumor, whereas coughing, hemoptysis, or shoulder pain may indicate a lung process (the so-called Pancoast tumor) and thus a second-order Horner syndrome. Other common causes of Horner syndrome among cancer patients are iatrogenic causes, such as surgery or radiation in the cervical and neck area. Heterochromia in children usually indicates congenital Horner syndrome and does not require extensive work-up or treatment. Pharmacologic testing with cocaine or apraclonidine may result in reversal of anisocoria in patients with Horner syndrome and can help in confirming the diagnosis; hydroxyamphetamine drops may also help localize the lesion [67].

Another cause of ptosis may be mechanical. For example, inflammatory changes in the upper eyelid due to orbital or paranasal sinus infection, surgical trauma, or external beam radiation therapy may cause temporary ptosis. An isolated tumor in the upper eyelid, such as a lacrimal gland carcinoma or lymphoma, plexiform neurofibroma, or any other tumor that extends to the superior orbit, may also lead to mechanical ptosis of the upper eyelid.

Flashes and Floaters

Flashes of light, "showers of new floaters," and a "curtain" coming down over the visual field can be ominous symptoms of vitreoretinal traction, possible retinal tear, or reti-

nal detachment (please refer to "acute vision loss" section mentioned earlier in the chapter). A thorough dilated funduscopic examination by an ophthalmologist is necessary to determine the exact nature of vitreoretinal pathology and to rule out retinal tears or retinal detachment in patients who complain of an acute onset of flashes and floaters, particularly if these symptoms are associated with a loss of vision.

Vitritis (inflammation of the vitreous gel) can be caused by intraocular neoplasms, most commonly leukemia and lymphoma, and it can present with the onset of floaters and gradual loss of vision [68]. In patients whom intraocular leukemia or lymphoma is suspected, a MRI of the brain/orbit with and without contrast is necessary to evaluate for central nervous system (CNS) involvement as the CNS is frequently involved [69]. A vitrectomy and vitreal biopsy may be required to make a definitive diagnosis [70].

A common benign condition that can also cause the acute onset of flashes and floaters is acute posterior vitreous detachment, which is mostly secondary to senescence or trauma [71, 72]. Various forms of retinitis, endogenous endophthalmitis, and posterior uveitis may also present with the same initial symptoms. Management of endophthalmitis includes blood cultures, possible vitreous biopsy to determine the causative organism, and immediate intravenous antibiotics.

Conclusion

Ophthalmological emergencies in cancer patients are multifaceted. In most instances, consultation with an ophthalmologist is necessary to insure the timely diagnosis and management of these conditions. A general understanding of the different components of an eye examination and the differential diagnosis for common ocular presentations may help the oncologist or emergency physician with triaging, work-up, and initial treatment of these conditions until the patient can be examined by an ophthalmologist.

References

- MacCumber MW. Management of ocular injuries and emergencies. Philadelphia: Lippincott, Williams & Wilkins; 1998.
- Rhee DJ. The Wills eye manual: office and emergency room diagnosis and treatment of eye disease. Philadelphia: Lippincott, Williams & Wilkins; 1999.
- Thompson HS, Corbett JJ, Cox TA. How to measure the relative afferent pupillary defect. Surv Ophthalmol. 1981;26(1):39–42.
- Liu GT, Volpe NJ, Galetta SJ. Neuro-ophthalmology: diagnosis and management. Philadelphia: W.B. Saunders; 2001.
- Esmaeli B, Medeiros LJ, Myers J, Champlin R, Singh S, Ginsberg L. Orbital mass secondary to precursor T-cell acute lymphoblastic leukemia: a rare presentation. Arch Ophthalmol. 2001;119(3):443–6.

- Landoy Z, Rotstein C, Shedd D. Aspergillosis of the nose and paranasal sinuses in neutropenic patients at an Oncology Center. Head Neck Surg. 1985;8(2):83–90.
- Fauci AS, Braunwald E, Isselbacher K, Wilson J, Martin J, Kasper D, et al. Harrison's principles of internal medicine. 14th ed. New York: McGraw-Hill (Health Professions Division); 1998. p. 120.
- Bianchi-Marzoli S, Brancato R. Tumors of the optic nerve and chiasm. Curr Opin Ophthalmol. 1994;5(6):11–7.
- Christmas NJ, Mead MD, Richardson EP, Albert DM. Secondary optic nerve tumors. Surv Ophthalmol. 1991;36(3):196–206.
- Mack HG, Jakobiec FA. Isolated metastases to the retina or optic nerve. Int Ophthalmol Clin. 1997;37(4):251–60.
- Albert DM, Jakobiec FA. Principles and practice of ophthalmology. Philadelphia: W.B. Saunders; 2000.
- Duker JS, Nielsen JC, Eagle RC Jr, Bosley TM, Granadier R, Benson WE. Rapidly progressive acute retinal necrosis secondary to herpes simplex virus, type 1. Ophthalmology. 1990;97(12):1638–43.
- Black RL, Terry JE. Ocular manifestations of thrombotic thrombocytopenic purpura. J Am Optom Assoc. 1991;62(6):457–61.
- Sallah S, Ahmad O, Kaiser HE. Pathogenesis of thrombotic disorders in patients with cancer. In Vivo. 2000;14(1):251–3.
- Kohner EM, Laatikainen L, Oughton J. The management of central retinal vein occlusion. Ophthalmology. 1983;90(5):484–7.
- Leigh RJ, Zee DS. The neurology of eye movement. Philadelphia: F.A. Davis; 1991. p. 227–31.
- Rizk SS, Kraus DH, Gerresheim G, Mudan S. Aggressive combination treatment for invasive fungal sinusitis in immunocompromised patients. Ear Nose Throat J. 2000;79(4):278–80. 282, 284–5
- Dawson CR, Hanna L, Wood TR, Despain R. Adenovirus type 8 keratoconjunctivitis in the United States. 3. Epidemiologic, clinical, and microbiologic features. Am J Ophthalmol. 1970;69(3):473–80.
- Kiang E, Tesavibul N, Yee R, Kellaway J, Przepiorka D. The use of topical cyclosporin A in ocular graft-versus-host-disease. Bone Marrow Transplant. 1998;22(2):147–51.
- Higa GM, Gockerman JP, Hunt AL, Jones MR, Horne BJ. The use of prophylactic eye drops during high-dose cytosine arabinoside therapy. Cancer. 1991;68(8):1691–3.
- Lass JH, Lazarus HM, Reed MD, Herzig RH. Topical corticosteroid therapy for corneal toxicity from systemically administered cytarabine. Am J Ophthalmol. 1982;94(5):617–21.
- Kartush JM, Linstrom CJ, McCann PM, Graham MD. Early gold weight eyelid implantation for facial paralysis. Otolaryngol Head Neck Surg. 1990;103(6):1016–23.
- Patipa M. Ophthalmic surgical management of facial paralysis. J Fla Med Assoc. 1990;77(9):839–42.
- Wilhelmus KR. The red eye. Infectious conjunctivitis, keratitis, endophthalmitis, and periocular cellulitis. Infect Dis Clin N Am. 1988;2(1):99–116.
- Ho JD, Xirasagar S, Lin HC. Increased risk of a cancer diagnosis after Herpes zoster ophthalmicus: a nationwide population-based study. Ophthalmology. 2011;18(6):1076–81.
- Mapstone R. Closed-angle glaucoma: theoretical considerations. Br J Ophthalmol. 1974;58(1):36–40.
- Reddy SC, Madhavan M, Mutum SS. Anterior uveal and episcleral metastases from carcinoma of the breast. Ophthalmologica. 2000;214(5):368–72.
- Lam DS, Lai JS, Tham CC. The management of acute angle-closure glaucoma. Eye (Lond). 2000;14(Part 3a):412.
- Dick AD. Immune mechanisms of uveitis: insights into disease pathogenesis and treatment. Int Ophthalmol Clin. 2000;40(2):1–18.
- Romero CF, Rai MK, Lowder CY, Adal KA. Endogenous endophthalmitis: case report and brief review. Am Fam Physician. 1999;60(2):510–4.

- Shen X, Gezhi X. Vitrectomy for endogenous fungal endophthalmitis. Ocul Immunol Inflamm. 2009;17(3):148–52.
- Keswani T, Ahuja V, Changulani M. Evaluation of outcome of various treatment methods for endogenous endophthalmitis. Indian J Med Sci. 2006;60(11):454.
- 33. Wobig JL, Wirta DL. Clinical and radiologic lacrimal testing in patients with epiphora. Ophthalmology. 1998;105(9):1574.
- El-Sawy T, Frank SJ, Hanna E, Sniegowski M, Lai SY, Nasser QJ, et al. Multidisciplinary management of lacrimal sac/nasolacrimal duct carcinomas. Ophthal Plast Reconstr Surg. 2013;29(6):454–7.
- Esmaeli B, Valero V. Epiphora and canalicular stenosis associated with adjuvant docetaxel in early breast cancer: is excessive tearing clinically important? J Clin Oncol. 2013;31(17):2076–7.
- Esmaeli B. Prospective study of incidence and severity of epiphora and canalicular stenosis in patients with metastatic breast cancer receiving docetaxel. J Clin Oncol. 2006;24(22):3619–22.
- Diba R, Saadati H, Esmaeli B. Outcomes of dacryocystorhinostomy in patients with head and neck tumors. Head Neck. 2005;27(1):72–5.
- El-Sawy T, Ali R, Nasser QJ, Esmaeli B. Outcomes of dacryocystorhinostomy in patients with head and neck cancer treated with high-dose radiation therapy. Ophthal Plast Reconstr Surg. 2012;28(3):196–8.
- Esmaeli B, Valero V, Ahmadi MA, Booser D. Canalicular fibrosis secondary to docetaxel (taxotere): a newly recognized side effect. Ophthalmology. 2001;108(5):994–5.
- Caravella LP Jr, Burns JA, Zangmeister M. Punctal-canalicular stenosis related to systemic fluorouracil therapy. Arch Ophthalmol. 1981;99(2):284–6.
- Esmaeli B, Golio D, Lubecki L, Ajani J. Canalicular and nasolacrimal duct blockage: an ocular side effect associated with the antineoplastic drug S-1. Am J Ophthalmol. 2005;140(2):325–7.
- Burns JA, Morgenstern KE, Cahill KV, Foster JA, Jhiang SM, Kloos RT. Nasolacrimal obstruction secondary to I(131) therapy. Ophthal Plast Reconstr Surg. 2004;20(2):126–9.
- 43. Shepler TR, Sherman SI, Faustina MM, Busaidy NL, Ahmadi MA, Esmaeli B. Nasolacrimal duct obstruction associated with radioactive iodine therapy for thyroid carcinoma. Ophthal Plast Reconstr Surg. 2003;19(6):479–81.
- Kloos RT. Nasolacrimal drainage system obstruction from radioactive iodine therapy for thyroid carcinoma. J Clin Endocrinol Metab. 2002;87(12):5817–20.
- Easty DL, Sparrow JM. Oxford textbook of ophthalmology. Oxford/New York: Oxford University Press; 1999.
- Cahill KV, Burns JA. Management of acute dacryocystitis in adults. Ophthal Plast Reconstr Surg. 1993;9(1):38–42.
- Margo CE, Mulla ZD. Malignant tumors of the orbit. Analysis of the Florida Cancer Registry. Ophthalmology. 1998;105(1):185–90.
- White WI, Ferry JA. Ocular adnexal lymphoma. A clinicopathologic study with identification of lymphomas of mucosa-associated lymphoid tissue type. Ophthalmology. 1995;102(12):1994–2006.
- 49. Esmaeli B, editor. Ophthalmic oncology. M.D. Anderson solid tumor oncology series, vol. 6. Boston: Springer; 2010.
- Esmaeli B, Amir Ahmadi M, Manning J, Mclaughlin PW, Ginsberg L. Clinical presentation and treatment of secondary orbital lymphoma. Ophthal Plast Reconstr Surg. 2002;18(4):247–53.
- 51. Woog JJ, Kim YD, Patrick Yeatts R, Kim S, Esmaeli B, Don K, et al. Natural killer/T-cell lymphoma with ocular and adnexal involvement. Ophthalmology. 2006;113(1):140–7.
- Sindhu K, Downie J, Ghabrial R, Martin F. Aetiology of childhood proptosis. J Paediatr Child Health. 1998;34(4):374–6.
- Weber AL, Romo LV, Sabates NR. Pseudotumor of the orbit. Clinical, pathologic, and radiologic evaluation. Radiol Clin North Am. 1999;37(1):151–68.
- Lessner A, Stern GA. Preseptal and orbital cellulitis. Infect Dis Clin N Am. 1992;6(4):933–52.

- Rumelt S, Rubin PA. Potential sources for orbital cellulitis. Int Ophthalmol Clin. 1996;36(3):207–21.
- Klapper SR, Lee AG, Patrinely JR. Orbital involvement in allergic fungal sinusitis. Ophthalmology. 1997;104(12):2094–100.
- Tole DM, Anderton LC, Hayward JM. Orbital cellulitis demands early recognition, urgent admission and aggressive management. J Accid Emerg Med. 1995;12(2):151–3.
- Hornblass A, Herschorn BJ, Stern K, Grimes C. Orbital abscess. Surv Ophthalmol. 1984;29(3):169–78.
- Dallow RL. Evaluation of unilateral exophthalmos with ultrasonography: analysis of 258 consecutive cases. Laryngoscope. 1975;85(11 Pt 1):1905–19.
- Goodall KL, Brahma A, Bates A, Leatherbarrow B. Lateral canthotomy and inferior cantholysis: an effective method of urgent orbital decompression for sight threatening acute retrobulbar haemorrhage. Injury. 1999;30(7):485–90.
- 61. Fleishman JA, Beck RW, Hoffman RO. Orbital emphysema as an ophthalmic emergency. Ophthalmology. 1984;91(11):1389–91.
- Hunts JH, Patrinely JR, Holds JB, Anderson RL. Orbital emphysema. Staging and acute management. Ophthalmology. 1994;101(5):960–6.
- 63. Muhammad JK, Simpson MT. Orbital emphysema and the medial orbital wall: a review of the literature with particular reference to that associated with indirect trauma and possible blindness. J Craniomaxillofac Surg. 1996;24(4):245–50.
- 64. Jacobson DM, Trobe JD. The emerging role of magnetic resonance angiography in the management of patients with third cranial nerve palsy. Am J Ophthalmol. 1999;128(1):94–6.
- Esmaeli B, Ginsberg L, Goepfert H, Deavers M. Squamous cell carcinoma with perineural invasion presenting as a Tolosa-Hunt-like

syndrome: a potential pitfall in diagnosis. Ophthal Plast Reconstr Surg. 2000;16(6):450–2.

- Bernard PD. Mitchell, Horner syndrome and others? Surg Neurol. 1979;11(3):221–3.
- Wilhelm H, Ochsner H, Kopycziok E, Trauzettel-Klosinski S, Schiefer U, Zrenner E. Horner's syndrome: a retrospective analysis of 90 cases and recommendations for clinical handling. Ger J Ophthalmol. 1992;1(2):96–102.
- Shields JA. Diagnosis and management of intraocular tumors. C.V. Mosby: St Louis; 1983.
- 69. Davis JL. Intraocular lymphoma: a clinical perspective. Eye. 2013;27(2):153–62.
- Rajagopal R, Harbour JW. Diagnostic testing and treatment choices in primary vitreoretinal lymphoma. Retina. 2011;31(3):435–40.
- Rothova A, Ooijman F, Kerkhoff F, Van Der Lelij A, Lokhorst HM. Uveitis masquerade syndromes. Ophthalmology. 2001; 108(2):386–99.
- Classe JG. Clinicolegal aspects of vitreous and retinal detachment. Optom Clin. 1992;2(3):113–25.
- Debnam JM. Neuroradiology of ocular and orbital tumors. In: Esmaeli B, editor. Ophthalmic oncology. M.D. Anderson solid tumor oncology series, vol. 6. Boston: Springer; 2010. p. 149–68. https://doi.org/10.1007/978-1-4419-0374-7_12.
- 74. Shinder R, Esmaeli B. Secondary orbital tumors extending from ocular or periorbital structures. In: Esmaeli B, editor. Ophthalmic oncology. M.D. Anderson solid tumor oncology series, vol. 6. Boston: Springer; 2010. p. 79–97. https://doi. org/10.1007/978-1-4419-0374-7_6.

Cardiovascular Oncologic Emergencies

Tarek Nafee, Dany Debs, Maria Isabel Camara Planek, Allison Zimmerman, Tawanna Charlton, Michelle Dimza, Alexandra Murphy, Stephen Dobbin, Ninian N. Lang, Sara Tyebally, Tarek Barbar, Syed S. Mahmood, Evelyn Horn, Umberto Campia, Arjun K. Ghosh, Ki Park, Michael G. Fradley, Rhian M. Touyz, Tochukwu M. Okwuosa, and Sherry-Ann Brown

Introduction

Cancer and its treatment are associated with an enormous burden for patients [1]. Cancer itself may be life-threatening, and cancer therapies are often associated with cardiovascular toxicities (Fig. 21.1) [2]. An overview of various cardiovascular toxicities will be presented, including ischemic manifestations, non-ischemic vascular conditions, pericardial diseases, cardiomyopathies, myocarditis, conduction disturbances, and valvular heart disease. We will also address the utility of imaging modalities and strategies in the prevention, diagnosis, and management of many of these emergent cardiovascular oncologic toxicities. Finally,

T. Nafee · D. Debs

M. I. C. Planek · A. Zimmerman · T. M. Okwuosa Department of Medicine, Rush University Medical Center, Chicago, IL, USA

T. Charlton Department of Medicine, Cardiology, College of Medicine, University of Florida, Gainesville, FL, USA

M. Dimza

Division of General Internal Medicine, Department of Medicine, Cardiology, University of Florida, Gainesville, FL, USA

A. Murphy

Department of Cardiology, Austin Health & The University of Melbourne, Melbourne, VIC, Australia

S. Dobbin · N. N. Lang Institute of Cardiovascular and Medical Sciences, University of Glasgow, Scotland, UK

S. Tyebally · A. K. Ghosh Barts Heart Centre, Barts NHS Health Trust, Department of Cardiology, St. Bartholomew's Hospital, London, England, UK

T. Barbar · S. S. Mahmood

Division of Cardiology, Department of Internal Medicine, New York-Presbyterian Hospital/Weill Cornell Medical Center, New York, NY, USA we end the chapter with brief comments on cardio-oncologic in the COVID-19 era.

Ischemic Cardiovascular Manifestations

Case Study

A 63-year-old postmenopausal female with a past medical history of untreated hyperlipidemia and obesity was diagnosed with invasive ductal carcinoma following screening mammogram. Approximately 18 months prior to this diagnosis, she presented with substernal chest pain and a computed

E. Horn

Department of Medicine, Cardiology Division, Perkin Center for Heart Failure, New York-Presbyterian Hospital/Weill Cornell Medical Center, New York, NY, USA

U. Campia

Division of Cardiovascular Medicine, Department of Medicine, Brigham and Women's Hospital, Harvard Medical School, Boston, MA, USA

K. Park

UF Women's Cardiovascular Health Sciences, Malcom Randall VAMC, Division of Cardiovascular Medicine, Department of Medicine, Cardiology, University of Florida, Gainesville, FL, USA

M. G. Fradley

Division of Cardiovascular Medicine, Department of Medicine, Perelman Center for Advanced Medicine, Perelman School of Medicine at the University of Pennsylvania, Philadelphia, PA, USA

R. M. Touyz

Institute of Cardiovascular and Medical Sciences, BHF Glasgow Cardiovascular Research Centre, University of Glasgow, Glasgow, Scotland, UK

S.-A. Brown (🖂)

Cardio-Oncology Program, Division of Cardiovascular Medicine, Department of Medicine, Medical College of Wisconsin, Milwaukee, WI, USA e-mail: shbrown@mcw.edu



Roger Williams Medical Center, Department of Medicine, Boston University School of Medicine, Providence, RI, USA



Fig. 21.1 Cardiovascular toxicities of various cancer therapies. (From Brown [2], Creative Commons Attribution License [CC BY])

tomography angiography (CTA) revealed proximal left anterior descending (LAD) coronary artery disease without flowlimiting stenosis, calcification, and possible flow-limiting stenosis within a diminutive diagonal branch with a calcium score of 45 and a Coronary Artery Disease-Reporting and Data System (CAD-RAS) score of 3. Initiation of aspirin and statin therapy were discussed at length; however, the patient declined. She underwent right breast mastectomy with pathology consistent with low-risk stage 1A ER+ HER2- RS2 oncotype. Per her primary oncologist, a plan was made to initiate aromatase inhibitor (AI) therapy without additional chemotherapy; however, initiation was delayed due to infection of the breast expander after reconstructive surgery.

On post-op day 1 of removal of the infected breast expander, and now 2 months from initial cancer diagnosis, the patient experienced sudden-onset chest pain and shortness of breath at home. EMS arrived within 30 minutes and administered 325 mg aspirin. En route, she went into ventricular fibrillation requiring cardiopulmonary resuscitation for approximately 1 minute and one dose of epinephrine after which return of spontaneous circulation was achieved. On arrival to the emergency department, she was hemodynamically stable, but in mild respiratory distress. Electrocardiogram demonstrated acute ST segment elevations in the anterior leads (Fig. 21.2). Bilevel positive airway pressure was started promptly, as well as a nitroglycerin infusion and administration of intravenous furosemide 80 mg. She was given 180 mg ticagrelor and taken emergently for cardiac catheterization.

Coronary angiography revealed 100% occluded proximal LAD without other significant coronary artery disease. Percutaneous coronary intervention (PCI) of the LAD was performed, and drug-eluting stent was deployed resulting in TIMI 3 flow. Transthoracic echocardiogram showed diffuse apical and mid septal left ventricular segment akinesis with mild reduction in ejection fraction of 40–45%. She was continued on dual antiplatelet therapy (DAPT) with aspirin and ticagrelor, initiated on guideline-directed medical therapy for heart failure and referred to cardiopulmonary rehabilitation with aggressive lifestyle modifications.

At outpatient follow-up, patient was doing well without cardiac symptoms. Prior to the myocardial infarction, the original plan from oncologic was to initiate letrozole, an AI, for adjuvant therapy. Alternative treatment with tamoxifen was considered, however felt to be suboptimal secondary to thrombotic risk and inferiority to letrozole for recurrence of breast cancer. After discussion with the patient's primary oncologist, the decision was made to defer initiation of letrozole for 12 months, while the risk of recurrent cardiovascular events was highest. Additionally, the decision was made to continue DAPT for an additional year following the initiation of letrozole. Atorvastatin was initiated, but hepatic enzymes became abnormal and thus the therapy was discontinued. This posed a significant challenge as it is well known



Fig. 21.2 12-lead ECG on presentation revealing anterior ST elevations

that letrozole (as well as other AIs) impacts lipid profiles. Alternative therapies such as evolocumab are being considered.

Case Discussion

We describe the case of a postmenopausal woman with recent diagnosis of ER+ HER2- breast cancer presenting with anterior STEMI and underwent PCI with drug-eluting stent placement. This is a patient with known nonobstructive coronary disease by CTA within a year prior to her MI event. Whether pre-existing atherosclerotic coronary disease contributed to this patient's presentation in the setting of heightened inflammation secondary to cancer is unclear. However, it is known that patients with cancer are at elevated risk of arterial thromboembolic (ATE) events manifesting as ischemic heart disease and stroke within the first several months after cancer diagnosis.

Following the management of her acute coronary syndrome, her treatment teams then needed to address longterm management of cancer and cardiovascular disease. Unfortunately, there are no current guidelines that address the vast majority of management decisions for a cancer patient experiencing an ATE event. There are infinite variables requiring risk and benefit analysis. For example, a large meta-analysis suggested the lack of survival benefit for AI over tamoxifen in breast cancer, even with longer cancerfree periods, may be secondary to the increased risk of cardiovascular events with AI. This elevated risk was amplified in women with pre-existing cardiovascular disease [3]. Furthermore, several studies have found an association of unfavorable changes in lipid profile for patients on letrozole vs tamoxifen [4, 5]. This data would suggest consideration of switching letrozole to tamoxifen for patients with elevated cardiovascular risk. However, others would argue that the overall risk of cardiovascular events in both letrozole and tamoxifen is consistently low in both trial arms, and this must be weighed against the elevated risk of venous thrombosis with tamoxifen as well as the established superiority of AI in decreasing cancer recurrence [3, 4, 6].

A common dilemma in the setting of myocardial infarction or thromboembolic stroke is selection of antiplatelet or anticoagulation regimen. The risk of thrombosis must be weighed against the propensity for bleeding in the setting of thrombocytopenia, platelet dysfunction, and anemia. The Society for Cardiovascular Angiography and Interventions published a consensus statement for cancer patients undergoing PCI, focusing mostly on thrombocytopenic patients with elevated bleeding risk. Recommendations included platelet cutoff values for each antiplatelet agent and dosing, as well as reduced duration for DAPT when platelet count <50,000 [7].

However, it is important to note that cancer patients have an increased risk of stent thrombosis, even on DAPT [8, 9]. Furthermore, several chemotherapy agents can negatively affect stent endothelialization secondary to the cytostatic and cytotoxic properties [10]. Vascular endothelial growth factor (VEGF) is an important molecule in negating restenosis, and therefore VEGF inhibitor therapies have been hypothesized to also increase risk of stent restenosis, although adequate data is lacking [10]. Ultimately, without large randomized controlled trials and data that can be easily extrapolated to a specific patient, management for acute ATE events in cancer patients will continue to require an individualized approach. For our patient, interdisciplinary discussion between cardiology, oncologic, and the patient was critical in complex decisionmaking. In this patient, given her low risk of cancer recurrence and low bleeding risk, a cautious approach of delaying AI for 12 months post-MI and extending DAPT for an additional 12 months was pursued.

Arterial Thrombosis

As described in the case, patients with underlying malignancies are at increased risk for arterial thrombosis. This risk is related to the frequent presence of a prothrombotic milieu in patients with cancer and/or to the direct adverse vascular and thrombogenic effects of various cancer treatments. Several mechanisms have been postulated to account for the occurrence of arterial thrombotic events, but the pathophysiology remains incompletely understood. However, it is appropriate to conceptually frame the discussion by separating the causes of arterial thrombosis secondary to the cancer itself from those associated with cancer treatment.

Secondary to Cancer

While the association between cancer and venous thromboembolism has been recognized for over a century, the increased risk of arterial thrombosis has been more recently appreciated. In particular, recent data from the Surveillance, Epidemiology, and End Results-Medicare database indicate a significant increase in arterial thromboembolism, including myocardial infarction [11]. These authors compared 279,719 patients with the most common types of malignancies in the United States (breast, lung, prostate, bladder, pancreatic, gastric, and colorectal cancer and non-Hodgkin lymphoma) with age- and sex-matched controls. At 6-month follow-up, patients with cancer had a substantially increased risk of arterial thromboembolism (composite of myocardial infarction and stroke) compared to controls: 4.7% (95% confidence interval [CI], 4.6% to 4.8%) versus 2.2% (95% CI, 2.1-2.2%) (hazard ratio, 2.2; 95% CI, 2.1-2.3), respectively. The risk was greatest in patients with lung cancer compared to other types of cancer and in patients with more advanced stages of cancer [12]. The hypercoagulable state observed in patients with cancer is likely mediated by a diverse array of mechanisms rather than a single common pathway. Cancerrelated factors include elevated neutrophil and platelet count, enhanced tissue factor expression, release of microparticles, and increased inflammatory mediators. Neutrophils can contribute to thrombus formation by extruding neutrophil extracellular traps (NETs), which consist of strands of nuclear chromatin, granular proteins, and proteases. NETs have been shown to directly activate the contact pathway, inhibit thrombomodulin, and promote red blood cell and platelet adhesion. In addition to high neutrophil count, patients with cancer frequently present with elevated platelet count, a known risk factor for VTE. Circulating microparticles are small spherical cellular fragments released from the plasma membrane of various cells. They may trigger thrombosis secondary to their high levels of tissue factor and by providing negatively charged phospholipids [13, 14].

Secondary to Cancer Therapy

Various cancer treatments are associated with an increased risk of venous and arterial thrombosis [14]. The mechanisms underlying this risk are complex and involve effects on the vascular endothelium, on the coagulation pathways, and on metabolic risk factors. Direct endothelial cell injury, such as from platinum-based agents, antimetabolites (5-FU), antimicrotubule agents (paclitaxel and vinblastine), and antitumor antibiotics (bleomycin), is likely the main mechanism whereby these agents increase thrombotic risk. In addition, VEGF inhibitors such as bevacizumab and VEGF-signaling pathway inhibitors such as sorafenib and sunitinib have also been associated with arterial thrombosis, probably related to their effects on endothelial function [15].

Premature Coronary Artery Disease

Secondary to Pharmacologic Therapy

While there is scant evidence that cancer therapy is directly a cause of atherosclerosis, several treatments have longlasting endothelial and metabolic effects that may predispose to the development of premature atherothrombosis. For example, patients who previously received cisplatin-based chemotherapy for testicular cancer had increased risk of developing lipid disorders, obesity, and hypertension [16].

As mentioned previously, AI therapy has been associated with adverse cardiovascular events, likely secondary to the impact on serum cholesterol. It is proposed that the decreased levels of estrogen during adjuvant hormonal therapy have downstream effects on HDL and LDL levels. Not all AI agents have the same effect, which may be explained by varying modes of action. Anastrozole and letrozole are nonsteroidal reversibly binding AIs. Letrozole has been associated with adverse effects on lipids, including increased LDL and total cholesterol. Anastrozole has little to no effect on lipids. By contrast, exemestane, a steroidal irreversibly binding AI, demonstrated a slight positive impact on lipids in clinical trials [17]. Of note, tamoxifen has estrogenic agonist effects on lipid profiles, with several studies showing a decrease in total and LDL cholesterol and mild increase in HDL [17, 18]. A recent large cohort-based retrospective analysis found an association of increased risk of myocardial infarction (HR = 2.08; 95% CI, 1.02, 4.27) but not cardiovascular mortality (HR = 0.87; 95% CI, 0.49, 1.54) when comparing patients switching from tamoxifen to AI vs continuing tamoxifen therapy [19]. Although there are no specific guidelines for management of hypercholesterolemia in this context, close lipid monitoring and identification of high-risk patients are crucial. Adherence to ACC/AHA guidelines is advisable for patients undergoing hormonal therapy to optimize cardiovascular risk management [20]. In special cases consideration of alternative hormonal therapy is reasonable.

Secondary to Radiation Therapy

Radiation therapy is an integral part of treatment in over 50% of patients with cancer and should be readily queried in the history obtained in the ED. Unfortunately, the effects of radiation are nonselective, and vessels within the field of radiation are vulnerable to acute injury, leading to endothelial dysfunction, inflammation, and thrombosis, as well as to long-term damage, with premature atherosclerosis and extensive calcification [21]. The clinical manifestations of radiation-induced vascular damage depend on the area of exposure. Radiation to the chest, such as for Hodgkin's lymphoma and breast cancer, has been associated with increased risk of premature CAD, whereas head and neck radiation confers a higher risk of carotid disease, transient ischemic attack, and ischemic stroke [22-26]. In support of this evidence, in 5-year survivors of childhood cancer, an average radiation dose exceeding 5 Gy is associated with a significant increase of risk of dying from cardiac disease [27]. Furthermore, patients who received mediastinal irradiation for Hodgkin's lymphoma had a 3.2-fold increased risk of requiring surgical coronary revascularization, a 1.6-fold risk of needing endovascular coronary procedure, and a 9.2-fold risk of requiring valve surgery [28]. Most modern radiotherapy programs have been able to deliberately exclude cardiac substructures from direct beam paths as part of the computerbased and image-guided treatment planning process.

Coronary Artery Vasospasm

The fluoropyrimidines antimetabolite 5-fluorouracil (5-FU) and its oral prodrug capecitabine are first-line agents for the treatment of head and neck, breast, and gastrointestinal cancers. One of the most concerning toxicities of 5-FU and capecitabine is the occurrence of coronary vasospasm, leading to ischemia, angina, myocardial infarction, arrhythmias, or sudden death. Postulated mechanisms of fluoropyrimidine-mediated vasospasm include endothelial dysfunction, with increased bioactivity of the potent vasoconstrictor endothelin-1, cytolysis with denudation of the intima, and enhanced smooth muscle cell contractility [14]. Coronary vasospasm

is a medical emergency and requires prompt medical attention and treatment. Given the primary role of fluoropyrimidines in the treatment of various malignancies, in patients who develop vasospasm, rechallenge may be considered with the use of cardioprotective pretreatment [29].

Vasculitis

The introduction of immunotherapies, which harness the body's own defense system to attack malignant cells, has represented an unprecedented leap forward in the treatment of a growing number of cancers. In particular, immune checkpoint inhibitors (ICIs) have shown high rates of sustained response in various malignancies and are now considered also as first-line therapy. However, the enhancement of immune responses has been associated with the occurrence of a wide range of autoimmune phenomena, with reported involvement of most organs. While less common than other manifestations, the involvement of the cardiovascular system has been of significant concern, as mortality in patients with ICI-mediated myocarditis is as high as 50%. In addition to myocarditis and pericarditis, vasculitides have also been reported in association with ICI. Any vascular bed can potentially be affected, but large vessel vasculitis of the aorta and temporal arteries are most commonly reported. As such, the ED practitioner should consider this as a possible etiology when evaluating known cancer patients for an acute presentation consistent with systemic vascular inflammation. Importantly, patients with temporal arteritis are at risk of permanent visual loss and require prompt treatment with highdose steroids [30].

Peripheral Artery Disease

Secondary to Pharmacologic Therapy

Tyrosine kinase and VEGF-signaling pathway inhibitors represent important therapeutic classes and have revolutionized management of several cancers. However, their use has been associated with unpredictable and at times severe vascular toxicity including hypertension, stroke, arterial thrombosis, and aortic dissection [14, 31]. Dasatinib has been linked to the development of pulmonary arterial hypertension, whereas nilotinib and ponatinib have been associated with an increased risk of myocardial ischemia, cerebrovascular accidents, and peripheral artery disease, in some cases requiring revascularization or leading to amputation. Of note, while the vascular toxicity of nilotinib may be related, at least in part, to its adverse effects on body weight and on glucose and lipid metabolism, the vascular toxicity of ponatinib appears to be a new form of microvascular angiopathy [32].

	Chemotherapeutic		Hypertension	
Drug class	agent	Clinical use	incidence (%)	VTE incidence (%)
Alkylating agents	Cisplatin	Bladder, breast, head and neck, lung, ovarian, testicular	14–53	8.5–12.9 (74% cases within first 2 cycles)
Histone deacetylase inhibitors	Vorinostat	Cutaneous T-cell lymphoma	-	4.7–8
Immunomodulators	Thalidomide	Multiple myeloma	-	<5-58
	Lenalidomide		-	3–75
VEGF-signaling pathway inhibitors	Bevacizumab	Colon, ovarian, cervical, lung, renal cell, hepatocellular, thyroid, sarcoma	26–55	8-14
	Axitinib		40	3
	Cabozantinib		22–37	4–6
	Lenvatinib		41-68	-
	Pazopanib		35–57	2-5
	Sorafenib		17–48	5.5
	Sunitinib		16–47	0.5–3
Tyrosine kinase inhibitors Bcr-Abl	Bosutinib	Chronic myeloid leukemia, sarcoma	7.8	-
	Dasatinib		<10	-
	Imatinib		4	-
	Ponatinib		67	5–27
	Nilotinib		8-10	-
EGFR	Erlotinib	Lung	-	1.2–11

Table 21.1 Chemotherapeutic agents associated with HTN and VTE and the incidence of each

Secondary to Radiation Therapy

As introduced above, radiation therapy induces extensive vascular wall injury. The biological consequences of this damage extend beyond the acute phase and lead to progressive abnormal remodeling, increased susceptibility to atherosclerosis (also in the absence of traditional risk atherosclerotic risk factors), and extensive calcific degeneration. The clinical manifestations usually reflect the occurrence of hemodynamically significant stenoses in various arteries (e.g., brachiocephalic, subclavian, renal, iliac) and the presence of severe calcifications (e.g., porcelain aorta with inability to perform coronary bypass surgery). Of note, the risk of radiation-induced vascular injury increases with young age at time of exposure, higher total cumulative dose, higher dose of radiation fractions, concomitant cancer treatment, and superficial location of the vessels [23–25].

Non-ischemic Vascular Conditions

Of the non-ischemic vascular disorders, hypertension and venous thromboembolism (VTE) are especially important, because (i) many antiangiogenic drugs, especially vascular endothelial growth factor inhibitors (VEGFIs), induce development of hypertension; (ii) some anticancer treatments promote thrombogenesis and coagulation; and (iii) the underlying cancer itself is a prothrombotic condition. VEGFI-induced hypertension develops rapidly, is often severe, and may be associated with posterior reversible leukoencephalopathy, a medical emergency [33]. Cancerassociated VTE is especially prevalent in metastatic disease and is often associated with comorbidities such as atrial fibrillation. An overview of classes of chemotherapy agents most commonly associated with the non-ischemic cardiovascular toxicity emergencies – hypertension and VTE – will be highlighted, including alkylating agents (e.g., cisplatin), histone deacetylase inhibitors (e.g., vorinostat), immunomodulators (e.g., thalidomide), and tyrosine kinase inhibitors (TKIs), which include inhibitors of Bcr-Abl, endothelial growth factor receptor (EGFR), and VEGF (Table 21.1). We will also address strategies in the prevention, diagnosis, and management of these conditions in the emergency setting.

Hypertension

Hypertension (HTN) is a common comorbidity in cancer patients (37%) [34]. Newer more-targeted cancer therapies such as VEGFI have been associated with an increased risk of HTN. Almost every clinical trial of VEGFI reports a treatment-associated rise in blood pressure. Furthermore, up to 80% of patients develop HTN, either de novo or worsening of previously controlled disease [35].

The clinical consequences of VEGFI-associated HTN can be severe. The rapid onset of HTN in patients not previously "conditioned" to high blood pressure can lead to end-organ complications, such as stroke, myocardial ischemia, heart failure, and acute kidney injury, and proteinuria. Furthermore, VEGFIs have been associated with the development of posterior reversible encephalopathy syndrome (PRES), characterized by headaches, confusion, visual impairment, and seizures [33]. These manifestations, however, may be reversible with prompt recognition and treatment of the severe hypertension.

Mechanisms underlying VEGFI-induced HTN are incompletely understood. However, VEGF inhibition leads to a reduction of vasodilator, nitric oxide (NO), and is associated with increased production of the vasoconstrictor, endothelin-1 [36, 37]. Other hypotheses include oxidative stress, capillary rarefaction, and activation of the reninangiotensin-aldosterone system (RAAS) [38]. Oxidative stress and excessive reactive oxygen species (ROS) production can lead to a reduction in NO bioavailability. Rarefaction is a reduction in density of arterioles and capillaries, and marked reductions in capillary density have been demonstrated with VEGF inhibition, although it is uncertain if this is a cause or effect of VEGFI-associated HTN. The role of RAAS activation is unclear as VEGFIs have been associated with a decrease in renin levels and angiotensin-converting enzyme (ACE) inhibition has a limited impact on VEGFIassociated HTN.

Diagnosis

The Cardiovascular Toxicities Panel of the National Cancer Institute recommends frequent monitoring of blood pressure, particularly during the first cycle of treatment. Home blood pressure monitoring and 24-hour ambulatory blood pressure monitoring should be pursued, where feasible, to diagnose and monitor HTN, with blood pressure \geq 140/90 mmHg defined as HTN [34].

Treatment

Prior to commencing cancer treatment, adequate blood pressure control (<140/90 mmHg) should be achieved in patients with pre-existing HTN. Patients with blood pressure ≥140/90 mmHg should be treated with antihypertensive therapy. VEGFI-induced HTN can be difficult to treat and may require multiple agents to achieve blood pressure control. The choice of antihypertensive agents generally follows national guidelines for first-line treatment of HTN, and there is no evidence of superiority of one agent over another. However, non-dihydropyridine calcium channel antagonists such as verapamil and diltiazem inhibit cytochrome P450 3A4 and should be avoided secondary to the potential for consequent VEGFI toxicity. ACE inhibitors or angiotensin receptor blockers (ARBs) may be of benefit in patients with left ventricular dysfunction or proteinuria. Other agents such as dihydropyridine calcium channel antagonists, betablockers, and diuretics can also be used. There is some evidence that phosphodiesterase inhibitors, NO donors, and endothelin-1 receptor antagonists may be effective; however, these have not been studied extensively in clinical trials.

End-organ complications of VEGFI-induced HTN should be managed as per existing guidelines. PRES is diagnosed with magnetic resonance imaging of the brain revealing characteristic posterior fossa changes on T2-weighted imaging reflecting edema. Treatment is largely supportive, with blood pressure control and withdrawal of VEGFI therapy and antiepileptic therapy to treat seizures.

Venous Thromboembolism

Cancer is a prothrombotic condition, and VTE risk appears to be greatest in those with distant metastases and established VTE risk factors. Independent risk factors include central venous catheters, immobility, heart failure, and chemotherapy [39].

The pathophysiological processes associated with VTE in cancer include secretion of inflammatory cytokines and interaction with endothelial cells and platelets with activation of clotting cascade, particularly via increased TF expression. Additionally, there is inhibition of anticoagulant factors such as activated protein C and thrombomodulin, as well as reduced fibrinolysis [40].

Cancer therapies can further contribute to the prothrombotic state from direct damage to the endothelium and cancer cells, inducing apoptosis and releasing inflammatory cytokines. Additionally, they increase the activity of procoagulants such as TF and reduce the production of anticoagulants such as protein C and antithrombin from direct hepatotoxicity [39]. Cisplatin-associated VTE may be induced by direct endothelial injury, vascular inflammation, hypomagnesemiainduced vasospasm, increased procoagulant activity, and reduced anticoagulation formation [41]. Thalidomide and its analogues increase levels of von Willebrand factor and, when used in combination with anthracyclines, have been shown to induce endothelial dysfunction and increase levels of TF [42]. TKIs have also been shown to increase levels of von Willebrand factor and TF reflecting activation of endothelial cells [43].

Diagnosis

DVT is diagnosed by compression Doppler ultrasonography which can readily be acquired in the ED, while the gold standard for PE diagnosis is spiral CTA of pulmonary vessels. Nuclear medicine techniques such as ventilation/perfusion scans and magnetic resonance imaging angiography are alternative options for those with contraindications to computed tomography (CT).

Prevention

Generally, thromboprophylaxis should not be offered to ambulant patients with cancer. However, it may be considered in high-risk patients and those with other risk factors for VTE such as obesity, previous VTE, central venous catheters, and immobility. The Khorana risk score is used to stratify the risk of VTE in cancer patients. This takes into consideration the type of cancer, platelet count, hemoglobin, and body mass index (BMI), with a score ≥ 3 classified as high risk [44]. Thromboprophylaxis with low molecular weight heparin (LMWH) or direct oral anticoagulants (DOACs) should be offered, taking bleeding risk and drug interactions into consideration. DOAC therapy has been associated with increased risk of bleeding with gastrointestinal and genitourinary malignancies. Patients treated with thalidomide and dexamethasone or anthracycline chemotherapy should also be considered for prophylaxis with aspirin (75-150 mg), low molecular weight heparin (LMWH), or direct oral anticoagulants (DOACs) [45].

Treatment

The American Society of Clinical Oncologic guidelines recommend treatment of cancer-associated VTE with LMWH or DOAC therapy. These are preferred over vitamin K antagonists secondary to improved efficacy. The American College of Emergency Physicians recommends treatment of VTE with vitamin K antagonists or LMWH, and the use of DOAC therapy as an alternative. However, no recommendations have been made specifically for cancer patients [46]. Anticoagulation treatment should continue for at least 6 months, with duration of treatment beyond this dependent on risks and benefits for individual patients.

Pericardial Diseases

Malignant involvement of the pericardium is detected in 1-20% of autopsy studies of cancer patients, and the spectrum of the acute clinical presentation can include acute pericarditis, pericardial effusion with or without cardiac tamponade, or effects of local cancer invasion (primary or secondary). Additionally, pericardial disease in cancer patients can be as a result of their cancer treatment, such as cytotoxic chemotherapy and/or radiotherapy. Determining the etiology and providing effective treatment can often be challenging. The natural history of pericardial diseases in the oncologic setting can be complicated with pericardial emergencies requiring prompt diagnosis, procedural intervention, intensive care with hemodynamic monitoring, and early aggressive management.

Pericarditis

Pericarditis is the commonest form of pericardial disease and is responsible for around 5% of presentations to emergency

departments with non-ischemic cardiac pain [47, 48]. Mortality rates can reach 1% [49].

Acute Pericarditis

Acute pericarditis can present a medical emergency manifesting with symptoms such as intense sharp chest pain, alongside other diagnostic signs such as pericardial friction rub, and widespread saddle-shaped or concave ST segment elevation on the electrocardiogram (ECG), with or without a pericardial effusion. At least two of the four aforementioned features should be present to make the diagnosis [50]. In an unselected cohort of pericarditis patients, approximately 5% of cases were attributable to underlying cancer [48]. This may develop via direct infiltration of malignant cancer cells via adjacent structures, pericardial hemorrhage, or hematogenous dissemination of cancer cells [51]. In addition, pericarditis may occur as part of the paraneoplastic syndrome. In some instances, it can be caused as an acute side effect of radiation therapy. This acute phase is mediated, in part, by tumor necrosis factor (TNF) and interleukins (IL) IL-1, IL-6, and IL-8, further leading to neutrophil infiltration [24].

Alongside markers of inflammation such as C-reactive protein and erythrocyte sedimentation rate (ESR), diagnosis of cancer as the cause of pericarditis requires imaging (e.g., CT scan or cardiac magnetic resonance (CMR) imaging), cytology of pericardial fluid, and, ultimately, biopsies confirming malignant infiltration within the pericardial tissue.

The treatment of acute pericarditis is usually as per recommended guidelines in the general population. However, it is to be recognized that many cancer patients may have a predisposition to bleeding due to abnormal blood counts or coagulation abnormalities secondary to their disease or treatment. It can thus be challenging to introduce routine therapy such as nonsteroidal anti-inflammatory agents in this context. As a result, there is often a greater and earlier use of other agents, e.g., colchicine and steroids, although this may not alter outcomes [49]. In the case of radiotherapy-induced acute pericarditis, treatment of primary malignancy should not be withheld because of this.

Immunotherapy-Induced Pericarditis

Adverse effects of ICIs are distinct from conventional cytotoxic chemotherapy and can be life-threatening if left unrecognized. Pericarditis is an infrequent cardiac toxicity of ICIs and can occur with coexisting myocarditis. A total of seven cases of ICI-associated pericardial disease have been described, two involved ipilimumab and five involved nivolumab [52]. The exact mechanisms underlying ICImediated pericardial disease remain unknown; however, they might include pericardial inflammation by ICI-stimulated cytotoxic T-cells.

The treatment regimens for ICI-associated pericarditis vary depending on the case; however, the principal strategy concentrates on targeting the hyperactive T-cell response with steroids often used as first-line agents.

Pericardial Effusion

Cardiac tamponade is the ultimate emergent complication of pericardial effusion, and is an absolute indication for urgent drainage, either by pericardiocentesis, a surgical pleuropericardial window, or by surgical pericardiotomy. Tamponade develops when the pericardial fluid pressure is greater than the intracardiac filling pressures, compromising the cardiac filling and leading to reduced cardiac output and cardiogenic shock. Patients may present acutely unwell with presyncopal symptoms, dyspnea, orthopnea, chest pain, and/ or arrhythmias. Clinical findings include sinus tachycardia, jugular venous distention, pulsus paradoxus (>10 mmHg drop in systolic blood pressure during inspiration), lower extremity edema, hypotension, and, eventually, circulatory collapse. ECG findings include low-voltage QRS complexes, electrical alternans (alternating amplitude or axis of the QRS complexes between each cardiac cycle), PR segment depression, and sinus tachycardia. Echocardiography is the diagnostic test of choice since it can help with establishing the diagnosis and also guide management.

Although malignant pericardial effusion can occur as an early manifestation, they are usually a late finding in patients with metastases. This condition usually occurs because of obstruction of lymphatic drainage or an excess of fluid secretion from tumor nodules on the pericardial surface. The majority of patients are asymptomatic, and the effusion is discovered incidentally on echocardiography ordered for other reasons during their treatment. The mainstay of treatment is to allow sufficient drainage of the pericardial fluid to relieve the symptoms and prevent recurrence. Pericardiocentesis is an easier and less invasive procedure than pericardial window surgery, allowing prompt treatment at the time of diagnosis. However, pericardiocentesis leads to a recurrence rate of up to 20% at 30 days, which is higher than recurrence after surgical drainage (1-10% of recurrence) [53]. Rarely, malignant pericardial effusions are managed with intrapericardial injection of chemotherapeutic agents.

Cytotoxic chemotherapy agents themselves are also associated with pericardial diseases. Traditional chemotherapy agents such as fludarabine, cytarabine, doxorubicin, docetaxel, and cyclophosphamide have been associated with acute pericarditis and pericardial effusion [54, 55]. Dasatinib, a TKI targeting BCR-ABL, has also been linked to an increased incidence of pericardial effusions.

In rarer cases, primary neoplasms of the pericardium such as malignant mesotheliomas and angiosarcomas can cause primary pericardial effusions via direct invasion and are associated with a poor prognosis. Specific to angiosarcomas, patients often present with pericardial effusions and/or in cardiac tamponade in about 56% of cases. Angiosarcomas are mostly resistant to chemotherapy and radiotherapy and necessitate surgical therapy.

Cardiomyopathies

Systolic Cardiomyopathy

Systolic cardiomyopathy (CM) has long been associated with traditional cytotoxic chemotherapeutic agent toxicity. In addition, thoracic radiation therapy (XRT), newer molecular-targeted agents, and immune checkpoint inhibitors (ICI) have been shown to produce left ventricular (LV) systolic CM (traditionally defined as reduction in LV ejection fraction [LVEF] by at least 10%, even if LVEF is >50%) in recent years. Systolic CM is divided into four stages: stage A is defined as exposure to a known cardiotoxin, stage B is asymptomatic LV dysfunction, and stages C and D are defined as symptomatic heart failure (HF) to end-stage HF [56]. The main agents causing systolic CM are summarized in Table 21.2 [25, 58-64]. Risk factors common to all classes include age (>65 or < 10 years old), uncontrolled hypertension, smoking exposure, pre-existing diabetes, and baseline systolic LV dysfunction (LVEF <50%) [56].

Stress Cardiomyopathy

Of the agents in Table 21.2, some have been shown to cause acute stress CM related to initial dose [58, 60]. This has especially been studied in anthracyclines. The incidence of stress CM in all cancer patients is up to 10%, which may be related to chemotherapy and XRT use [61]. Cyclophosphamide >140 mg/kg or >1.55 g/m²/day is known to cause acute systolic CM by mechanism of acute myopericarditis, which can be hemorrhagic. Rituximab has been shown to cause acute infusion-related cardiogenic shock [61]. Sunitinib and sorafenib can cause stress CM with stress-induced angiogenesis [64]. Antimetabolites such as 5-fluorouracil; clofarabine and capecitabine; lomustine + bevacizumab + cyclophosphamide + etoposide; rapamycin; interferon alpha; and trastuzumab have demonstrated an ability to cause cardiogenic shock secondary to stress CM, requiring ionotropic agents in up to 20% of patients [61].

Immune Checkpoint Inhibitor-Induced Cardiomyopathy

Adverse cardiac events associated with ICI therapy are rare (less than 1% of patients) but can be potentially life-threatening. The most common and well characterized is

 Table 21.2
 Systolic cardiomyopathies

Class	Agents and incidence [58, 61]	Mechanism [58–61, 63]	Class-specific risk factors ^a [25, 57, 60, 62, 64]
Anthracyclines and analogues	Doxorubicin Epirubicin Idarubicin Mitoxantrone Class Inc: 3–48%	Multiple-stress theory: ROS via iron chelation Nucleus topoisomerase IIB activation Iron overload Inhibition of DNA repair	Cumulative dose: Doxorubicin >360 mg/m ² Daunorubicin >800 mg/m ² Epirubicin >720 mg/m ² Idarubicin >150 mg/m ² Mitoxantrone >160 mg/m ² Concomitant XRT Renal dysfunction, female sex
Monoclonal antibodies	Trastuzumab (HER2 target): 1–20%	Interruption of ErbB signaling, upregulated by ROS	Concomitant or prior anthracycline use
Alkylating agents	Cisplatin Cyclophosphamide Ifosfamide Class Inc: 7–38%	Direct endothelial and DNA injury Lipid peroxidation ROS Hemorrhagic myopericarditis	Usually during initial dose Dose-related (>140 mg/kg), irreversible in up to 25% at dose >1.55 g/m ² /day
Antimetabolites	5-Fluorouracil: 1–68% Capecitabine: 5.5%	Apoptosis of myocardial and endocardial cells Myocarditis Angiogenesis inhibitors Coronary vasospasm Cardiac arrhythmias	Usually during drug infusion Concomitant cisplatin use
Antimicrotubules	Paclitaxel Docetaxel Class Inc: 2–13%	NCS-1 activation \rightarrow calcium overload	Concomitant anthracycline use
Proteasome inhibitors	Bortezomib: 25–% Carfilzomib: 42–8%	NF- $\kappa\beta$ suppression	Baseline subclinical heart disease, measured by GLS
Protein kinase inhibitors, with tyrosine kinase inhibitors	Dasatinib: 3–29% Lapatinib, imatinib mesylate, cobimetinib, vemurafenib, trametinib, ibrutinib Inc: <1%	Mitochondrial damage Intrinsic apoptotic pathway	a
Vascular endothelial growth factor receptor inhibitors (VEGFR TKIs)	Sorafenib Sunitinib Lenvatinib Pazopanib Regorafenib Class Inc: 1.5–14%	Inhibits platelet-derived growth factor Inhibits nitric oxide generation → myocardial stress	Direct myocardial toxicity amplified by hypertension
Monoclonal antibody VEGFR	Bevacizumab Class Inc: 2–4%	Activated tissue factor release Prevention of endothelial regeneration	History of arterial thrombotic events
Immune checkpoint inhibitors	Nivolumab Pembrolizumab Ipilimumab Atezolizumab Durvalumab Class Inc of myocarditis 0.27%–1.14%	Out of those who develop myocarditis, ~50% present with systolic cardiomyopathy	Combination of ICIs ^b
Radiation therapy	N/A	Endothelial damage Acceleration of CAD ROS Fibrosis Sustained inflammation via NF+6	Age at XRT (>50 y); anterior or left chest XRT; concomitant chemotherapy with anthracyclines; high cumulative dose of XRT fractions (>2 Gy/ day) or total dose >30 Gy; pre-existing CAD

CAD coronary artery disease, *ErbB/HER2* human epidermal growth factor receptor 2, *GLS* global longitudinal strain, *Gy* Gray, *ICIs* immune checkpoint inhibitors, *Inc* incidence, *NCS-1* neuronal calcium sensor 1, *NF*- $\kappa\beta$ nuclear factor j-light-chain-enhancer of activated B-cells, *ROS* reactive oxygen species, *TKIs* tyrosine kinase inhibitors, *VEGFR* vascular endothelial growth factor receptor, *XRT* radiation therapy

^aRisk factors common to all classes include age (>65 or <10-year-old), uncontrolled hypertension, smoking exposure, pre-existing diabetes, and baseline systolic LV dysfunction (LVEF <50%) [56]

^bNote early versus late ICI toxicity described in text
ICI-induced acute fulminant myocarditis which typically occurs within 90 days of initiation of therapy, and is often associated with acute hemodynamic failure and death in up to 50% of patients [63]. The most commonly used and studied are nivolumab, pembrolizumab, and ipilimumab. Animal models suggest cardiac dysfunction may occur secondary to autoimmune response to cardiac troponin-I in PD-1-deficient mice via stimulation of calcium ion influx [65]. Fulminant myocarditis is more common in those receiving combination therapy ICI compared to monotherapy and is also associated with higher mortality rates [63].

Late ICI cardiac adverse effects have just begun to be examined. In a recent retrospective cohort study, late ICI cases (defined as >90 days after therapy initiation) revealed significantly more LV systolic dysfunction in 73.7% of cases (described as LVEF drop of >10% or LVEF <50%, with or without HF symptoms) than early ICI cases, and significantly higher rates of symptomatic HF (47% versus 21%, respectively; p = 0.01) [66]. Though overall mortality rates were not different between the two groups, it is important to be aware of the two clinically separate entities of ICI-induced cardiotoxicity.

Diastolic and Restrictive Cardiomyopathy

Radiation Therapy

XRT may cause systolic or diastolic dysfunction, with diastolic dysfunction being more common [25]. One singleinstitution study found that of 294 asymptomatic patients with Hodgkin's lymphoma treated with mediastinal XRT, 14% developed some form of diastolic dysfunction during the mean follow-up period of 3.4 years [67]. Factors that increase the risk of XRT-induced injury are shown in Table 21.2.

Acutely, XRT-induced inflammation results in mild myocardial dysfunction. Between 5 and 30 years post XRT, myocardial edema and inflammation directly lead to non-ischemic myocardial fibrosis [26]. The rate of ischemic heart disease, including myocardial infarction, is known to increase after mediastinal XRT [59, 68]. Myocardial ischemia can compound concurrent myocardial dysfunction secondary to XRT-induced myocardial fibrosis, potentiating the risk for flash pulmonary edema and acute progression to advanced HF [67]. XRT may lead to restrictive CM several years to decades after exposure [61].

In the Framingham study, systolic and diastolic dysfunctions after XRT were found in 13%, 18%, and 29% of patients with latency periods of 2–10, 11–20, and >20 years, respectively, compared to age-matched controls without XRT [60]. Authors did not assess for prior anthracycline use [21].

Anthracyclines and Monoclonal Antibodies

The effects of chemotherapy on diastolic dysfunction remain incompletely defined. In a recent longitudinal cohort study of 364 breast cancer patients, doxorubicin or doxorubicin followed by trastuzumab therapy was shown to produce diastolic dysfunction at 60% in 1 year, 70% by 2 years, and 80% by 3 years, with diastolic dysfunction shown to correlate with reduction in LVEF (2.1%; p < 0.001), and global longitudinal strain (GLS) (0.6%; p 0.013) [69]. Trastuzumab alone was not associated with diastolic dysfunction. Further studies are needed to determine long-term risk of HF in this population [69].

Cardiomyopathy Diagnosis

Cardiac dysfunction in oncologic patients and survivors is diagnosed by careful monitoring of the LV. Baseline LVEF and quantitative evaluation of diastolic dysfunction should be measured before treatment [56]. GLS should be measured in those with baseline GLS monitoring [56, 62]. Though routine use of cardiac biomarkers to detect CM is not wellestablished, some biomarkers such as troponin levels have been shown to predict CM in high-risk patients, such as those with pre-existing cardiovascular disease and those receiving high-dose anthracyclines [56]. In the most recent largest prospective study of patients treated with anthracyclines and/or trastuzumab, high sensitivity cardiac troponin T > 14 ng/L at completion was associated with a twofold risk of CM, N-terminal pro-B-type natriuretic peptide (NT-proBNP) was shown to predict CM in sequential anthracycline-trastuzumab therapy, and myeloperoxidase was shown to be a promising new biomarker for anthracyclineinduced CM [70].

For CM associated with ICI, early recognition and diagnosis of ICI-related myocarditis are critical secondary to its life-threatening potential. Shortness of breath is the most common presenting symptom [63]. Cardiac troponin appears to be a valid marker with a sensitivity of 94–100% [71]. Elevated NT-proBNP, peripheral edema, angina, and electrocardiogram abnormalities are common findings [63]. Severe conduction diseases including ventricular tachycardia and complete heart block are frequently seen.

Early detection in cardiogenic shock can be difficult as most patients have preserved LVEF, which can make diagnosis challenging. Whereas reduced LVEF may be a late finding, abnormal GLS may detect early evidence of cardiotoxicity not readily diagnosed by 2D echocardiography and can be a useful screening tool [62].

Cardiomyopathy Treatment

Patients who develop systolic LV dysfunction (LVEF <50%) or in stage B during anthracycline treatment should be treated with HF goal-directed medical therapy, including betablockers, angiotensin-converting enzyme inhibitors, or angiotensin II receptor blockers [56, 57]. The most studied agents are carvedilol, metoprolol, and bisoprolol, with notable recent trials showing significant reduction of mean change in LVEF using candesartan versus metoprolol and placebo. Statins, aspirin, enoxaparin, and rivaroxaban/apixaban have some clinical trial data to suggest cardioprotection as well [56]. Other cardioprotective strategies for anthracyclines include choosing liposomal formulations, using less cardiotoxic ones - epirubicin, idarubicin, and mitoxantrone - and continuous rather than bolus administration [56, 57]. Dexrazoxane, a cardioprotective agent via iron chelation and topoisomerase II inhibition, may be used if cumulative doxorubicin is >300 mg/ m² and/or the disease is metastatic [72]. In stage C and D HF, critical decision on withholding further chemotherapy, implantable cardioverter-defibrillator placement, cardiac resynchronization therapy, and more advanced measures of inotropic support to transplant are considered [56, 60].

For ICI, by recent guidelines, it is recommended to discontinue ICI therapy with any sign of cardiotoxicity [56, 73]. In patients with severe disease with hemodynamic compromise, intravenous prednisone 1–2 mg/kg, or methylprednisolone 500–1000 mg/kg, is recommended as first-line therapy. Addition of mycophenolate or infliximab can be considered in patients with suboptimal response to corticosteroids [74]. In corticosteroid-refractory cases, the use of antithymocyte globulin (ATG) and the combination of alemtuzumab and abatacept have been found to be effective [75, 76].

In cases of cardiogenic shock, treatment overall does not differ from that of the general population [61]. Ensuring hemodynamic stability is the first priority, and advanced mechanical circulatory support with extracorporeal membrane oxygenation or ventricular assist devices can be considered for patients with refractory cardiogenic shock and good life expectancy [61]. Treatment options for cardiogenic shock in the setting of fulminant XRT-induced myocarditis include immunoglobulin, ATG, tacrolimus, mycophenolate, and infliximab [61].

Ultimately, prevention of cardiotoxicity at the nonpharmacologic level should start before cancer treatment begins (before stage A), and with the help of a multidisciplinary team involving oncologic, cardiology, and immunology.

Myocarditis

ICIs are antibodies that enhance antitumor immunity by increasing the activation of T-cells [63, 77, 78]. Since the US Food and Drug Administration (FDA) approved the first ICI

in 2011, the application of this immunotherapy has increased from late-stage malignancy to adjuvant and even first-line settings in early malignancy [77, 78]. The number of ICI therapies continues to grow. Nivolumab, pembrolizumab, and cemiplimab target the programmed cell death receptor-1 (PD-1); ipilimumab and tremelimumab target the cytotoxic T-lymphocyte antigen-4 (CTLA-4); and atezolizumab, avelumab, and durvalumab target the programmed cell death ligand-1 (PDL-1) [77]. Consequences of increased immune system activity include disrupted immunological homeostasis and resultant inflammatory disease of nearly every organ system, e.g., the gastrointestinal tract, endocrine glands, skin, muscle, and liver, with complications termed immunerelated adverse events (irAEs) [77]. The severity of irAEs varies, based on the organ system involved. The spectrum of cardiovascular toxicity can include myocarditis, pericarditis, and vasculitis, among other conditions. Myocarditis in particular is associated with an elevated case fatality rate of up to 30–50% [63, 79].

Presentation

Data on myocarditis following ICI therapy are limited by their retrospective nature, given the relatively low prevalence of this irAE at up to 1.14% [63, 80]. Awareness of myocarditis as an irAE has been rising, with an increase in reporting incidence over time [81]. The median time from ICI initiation to myocarditis admission is 51 days [82], with 4 out of 5 patients presenting within the first 3 months [63]. Symptoms include chest pain, shortness of breath, fatigue, orthopnea, and paroxysmal nocturnal dyspnea [63]. Myocarditis can often occur with concomitant myositis, hepatitis, and myasthenia gravis [81]. In our experience, electrocardiographic changes include PR interval prolongation, QRS interval prolongation in a right bundle branch pattern progressing to high-grade atrioventricular conduction block and ventricular tachycardia. Troponin has been elevated in nearly 90% of cases and fourth-generation troponin T assay ≥ 1.5 ng/mL is associated with a fourfold increase in major adverse cardiovascular events (MACE; defined as a composite of cardiovascular death, cardiogenic shock, cardiac arrest, and hemodynamically significant complete heart block) [63]. LVEF measured by transthoracic echocardiogram (TTE) or cardiac magnetic resonance (CMR) is preserved in over one half of cases at presentation [63, 78]. However, subclinical left ventricular dysfunction in myocarditis patients is evident in the form of reduced GLS on TTE, such that myocarditis patients have a median 14% GLS, compared to normal GLS among controls without myocarditis. Furthermore, each percentage decrease in GLS is associated with a 1.5-fold increase in MACE among patients with a reduced EF and a 4.4-fold increase in MACE among patients with normal LVEF; GLS ≤14% is associated with the highest major adverse cardiovascular event (MACE) rate. GLS has been predictive of a nearly twofold increase rate of MACE in a multivariable model adjusted for EF [82].

Diagnosis

Suspected myocarditis patients with rising troponin should undergo emergent left heart cardiac catheterization, with left ventriculogram if possible, to rule out acute coronary syndrome and stress cardiomyopathy. If cardiac catheterization demonstrates nonobstructive coronary artery disease, we favor right ventricular endomyocardial biopsy as the gold standard for diagnosis of myocarditis, as a diagnosis of ICIassociated myocarditis could preclude further possibly lifeprolonging ICI cancer therapy. Endomyocardial biopsy in myocarditis demonstrates a relatively distinctive diffuse lymphohistiocytic infiltration characterized by a predominant CD163-positive histiocytic infiltrate, with an associated CD8+ and PD-1+ T-lymphocytic infiltrate, some of which is granzyme B-positive [83]. In the acute setting, use of CMR for myocarditis diagnosis appears limited, as only one in five myocarditis cases demonstrates late gadolinium enhancement (LGE) when the CMR is performed at <4 days from myocarditis presentation; it may take some time for myocardial fibrosis to develop and accumulate before becoming detectable on CMR [78]. On the other hand, when CMR is performed later in the clinical course of myocarditis $(\geq 4 \text{ days})$, nearly three out of four myocarditis patients have LGE [78]. Nevertheless, even among patients with myocarditis diagnosed with histopathology showing lymphocytic infiltration, only 38% have LGE and only 26% have an elevated T2-weighted STIR signal on CMR [78]. MACE does not necessarily associate with the presence of an LGE pattern or elevated T2-weighted STIR signal. Overall, as earlier treatment of myocarditis is associated with a lower rate of MACE, endomyocardial biopsy should still be considered among patients with a negative CMR, if there is a strong clinical suspicion for myocarditis.

Treatment

Myocarditis resulting from ICI cancer therapy should warrant rapid assessment, similar to acute coronary syndrome, as a delay is associated with a higher rate of MACE [63]. In suspected myocarditis cases, early initiation of corticosteroids in <24 hours from presentation has been associated with a 7% MACE rate, whereas corticosteroid initiation at >72 hours has associated with an 85% MACE rate [84]. There was an inverse relationship between initial dose of corticosteroids (measured in methylprednisolone equivalents) and the occurrence of MACE where <60 mg/day recipients experienced a 62% MACE rate, while 501–1000 mg/day recipients experienced a MACE rate of only 22% [84]. Overall, initiation time of corticosteroids appears to play an important role such that using a higher dose of corticosteroids may not overcome the late effect of delayed corticosteroids [84]. We favor treating suspected myocarditis cases, including those with pending endomyocardial histopathology results, with 1000 mg/day of methylprednisolone, which in our experience is associated with decreasing troponin levels and improvement in cardiomyopathy and tachybrady arrhythmias within 24 hours of corticosteroid initiation.

When needed, we utilize mechanical circulatory support, such as intra-aortic balloon pump or Impella, a micro-axial flow pump placed across the aortic valve, in myocarditis patients experiencing cardiogenic shock, with concurrent initiation of immunosuppressive therapy. Whether to start extracorporeal membrane oxygenation should involve shared decision-making among the providers and the patient or patient surrogates while keeping in mind long-term cancerrelated prognosis.

We dose 1000 mg/day of methylprednisolone for 3–5 days [85], followed by prednisone taper starting at 1 mg/kg/ day, provided troponin assays show a downward trend, and any complete heart block, ventricular arrhythmias, or cardiogenic shock has resolved [63]. Clinicians can also consider starting mycophenolate as a steroid-sparing agent to facilitate a more rapid taper off corticosteroids, although robust data on use of mycophenolate in this clinical context is lacking.

There is limited data on second-line immunosuppressive agents for patients refractory to corticosteroids with most case report-based data available for ATG therapy [75]. In a single-patient case report, the CTLA-4 agonist abatacept has been reported to be effective for steroid-refractory myocarditis [86]. However, cancer outcomes after CTLA-4 agonist therapy remain unknown. In another single-patient case report, alemtuzumab, which is an anti-CD52 monoclonal antibody, has also been shown to treat steroid-refractory myocarditis; the safety of alemtuzumab use in this population warrants further investigation [87]. There is paucity of non-case report-based data on the use of other therapies such as infliximab, intravenous immunoglobulin, and plasma exchange.

Conduction Disturbances

Arrhythmias are an increasingly recognized issue for cancer patients. Heart rhythm disturbances can be a consequence of various cancer therapeutics and can significantly impact optimal delivery of care. The majority of observed arrhythmias are supraventricular, particularly atrial fibrillation, and can lead to significant morbidity but are rarely life-threatening [88]. If atrial fibrillation or other supraventricular tachycardia is accompanied by severe hypotension, then emergent cardioversion is necessary; however, most cases can be managed conservatively [89]. Less commonly observed arrhythmic complications include ventricular tachyarrhythmias and conduction disorders such as complete heart block. These can be either a direct consequence of the cancer therapy or a result of another cardiotoxicity such as cardiomyopathy or ischemia. Regardless, when these occur, expeditious intervention is necessary to avoid potentially serious and life-threatening consequences.

The risk of ventricular arrhythmias in cancer patients is primarily related to QT prolongation, which is frequently encountered in this population. The QT interval is an ECG measure of ventricular depolarization and repolarization, and significant prolongation is a marker of increased risk of torsades de pointes (TdP), a specific type of polymorphic ventricular tachycardia (VT) [90]. The likelihood of developing TdP increases with QT intervals greater than 500 milliseconds (ms) or with a change in the QT interval of more than 60 ms from baseline [91]. Among cancer patients, various factors contribute to QT prolongation. Specifically, electrolyte abnormalities such as hypokalemia and hypomagnesemia are well known to prolong the QT interval and lead to TdP, and these frequently occur secondary to cancer-associated nausea, vomiting, and diarrhea. Moreover, QT prolongation is more common in older individuals, women, as well as patients with underlying renal and cardiovascular disease [91]. Additionally, various pharmaceuticals frequently prescribed to cancer patients have QT-prolonging potential which can be augmented in the setting of the aforementioned conditions. Specifically, certain antibiotics, antifungals, antidepressants, and anti-nausea medications are well known to increase the QT interval [91]. Additionally, various cancer therapeutics from traditional cytotoxic chemotherapy to targeted therapies are associated with QT prolongation [92]. Despite significant attention paid to the QT interval in oncologic drug development and monitoring, the actual risk of arrhythmia remains quite low. A study by MD Anderson evaluating patients enrolled in various phase 1 clinical trials reported a 20% incidence of QT prolongation with rates of TdP well below 1% [93]. Electrolyte repletion, avoidance of multiple QT-prolonging agents, and routine ECG monitoring are common strategies to mitigate the risk.

Among cytotoxic chemotherapeutics, QT prolongation is most frequently observed with arsenic trioxide, which is used primarily in the treatment of acute promyelocytic leukemia. In one study of over 3000 ECGs, 26% had a QT interval of more than 500 ms though no clinically significant arrhythmias were reported [94]. Implementation of aggressive risk mitigation strategies including electrolyte repletion and discontinuation of therapy when the QT interval is >500 ms, with resumption of the drug once the QT interval is <460 ms, has led to low TdP event rates [94]. Nevertheless, this drug carries a FDA black box warning for QT prolongation and sudden cardiac death [95].

QT interval prolongation is also an issue with certain targeted therapeutics including TKIs and cyclin-dependent kinase (CDK) 4/6 inhibitors. Among the TKIs, at least ten have standard or black box warnings for QT prolongation and sudden cardiac death. TKIs most commonly associated with QT prolongation include sunitinib, nilotinib, and vandetanib, with the latter two having an FDA black box warning. Sunitinib is a small molecule TKI used in the treatment of various solid tumors including renal cell carcinoma and gastrointestinal stromal tumors (GIST). There was a 2.3% incidence of QT prolongation greater than 500 ms with sunitinib though malignant arrhythmias occurred in less than 0.1% of treated patients [96]. Nilotinib is a secondgeneration BCR-ABL TKI used in the treatment of chronic myeloid leukemia. In the ENESTnd trial, a phase 3, randomized, open-label study, no patients developed a Fridericia-corrected QT interval of more than 500 ms [97]; however, sudden cardiac death occurred in 0.3% of patients [98]. Although preceding QT prolongation was not documented in any of these cases, the drug still carries a black box warning [98]. Vandetanib is a TKI used for medullary thyroid cancer also which carries a black block warning; however, OT interval prolongation is substantial (up to 18%) and arrhythmic complications have been observed [99]. Nevertheless, while a recent meta-analysis confirmed the QT-prolonging potential of these drugs, it was not predictive of TdP or other life-threating arrhythmias [100]. Finally, QT prolongation is also a significant cardiotoxicity of ribociclib, a CDK 4/6 inhibitor used to treat metastatic hormone receptor-positive, HER2-negative breast cancer. In the MONALEESA-2 trial, 3% of participants had an increase in their baseline QT interval by >60 ms, with 3.6% having an average QT interval of >480 ms [101]. Although no episodes of TdP have been definitely proven, it is recommended to avoid using ribociclib if the baseline QT interval is >450 ms and hold the drug if the QTc prolongs to a value >480 ms [102]. Interestingly, this does not appear to be a class-related effect as other CDK 4/6 inhibitors do not demonstrate similar QT-prolonging potential.

Several cancer therapeutics are directly arrhythmogenic, leading to malignant ventricular arrhythmias in the absence of other cardiovascular abnormalities or QT prolongation. Ibrutinib is a small molecule inhibitor of the Bruton's tyrosine kinase used to treat various B-cell malignancies. While ibrutinib is most associated with atrial arrhythmias [103], there is also an increased risk for ventricular arrhythmias with an incidence of 596 per 100,000 person years [104]. The mechanism of ibrutinib's arrhythmogenesis remains unclear; however, there is increased mortality in the setting of ibrutinib-associated arrhythmias [105].

Life-threatening ventricular arrhythmias can also occur as a consequence of another cardiotoxicity. They are frequent complication ICIs, which harness the body's immune system to target cancer cells. Ventricular arrhythmias most commonly occur in the setting of myocarditis; however, they can also occur in the absence of documented myocardial inflammation and may be the first manifestation of ICI cardiotoxicity [106]. Overall incidence of ICI-induced ventricular arrhythmias is reported at 27%, with an incidence of 7% in the setting of normal EF [107]. In addition, ventricular arrhythmias also occur in the setting of anthracyclineinduced cardiomyopathy. Several studies have demonstrated rates of ventricular arrhythmias similar to other non-ischemic cardiomyopathy etiologies. In a study by Mazur and colleagues, incidence of non-sustained VT was 73.9%, with sustained VT or ventricular fibrillation (VF) at 30.4% [108]. In a related study, the incidence of non-sustained VT, sustained VT, or VF was reported at 44.4% [109]. Ventricular arrhythmias may also occur as a result of myocardial ischemia/infarction which can be seen in the setting of 5-fluorouracil (5-FU)-induced coronary vasospasm [110].

If patients develop hemodynamically unstable ventricular arrhythmias associated with cancer therapeutics, treatment should be initiated using standard advanced cardiac life support (ACLS) algorithms, with prompt CPR and defibrillation if the patient is unconscious and without pulse or cardiovascular perfusion [111]. If the drug is actively being delivered via infusion, as is the case with ICIs or 5-FU, the treatment should be immediately stopped. For patients experiencing TdP, electrolytes should be promptly repleted, with intravenous magnesium given regardless of plasma concentrations [111]. Cardiac pacing is also an option since the QT interval shortens at higher heart rates (generally greater than 100 beats per minute) [112]. For patients with chemotherapyinduced cardiomyopathy and persistent systolic dysfunction, implantable cardioverter-defibrillators (ICDs) can be considered depending on the patient's oncologist prognosis and overall goals of care [112]. In fact, the MADIT-CHIC study demonstrated significant improvement in EF and LV volume in patients with chemotherapy-induced cardiomyopathy (EF <35%) and concomitant left bundle branch block [112]. In general, future administration of the offending cancer therapeutic should be avoided in patients that have developed a life-threatening ventricular arrhythmia except in rare and individualized circumstances.

Bradyarrhythmias are associated with various cancer therapeutics including the taxanes and crizotinib, an ALKinhibitor TKI for non-small lung cancer; however, they are generally asymptomatic and inconsequential [113, 114]. Rarely, severe symptomatic bradycardia or advanced conduction system disease including complete heart block can occur with certain treatments. High-degree atrioventricular block is a known manifestation of ICI cardiotoxicity [80]. In a large multicenter ICI myocarditis registry, complete heart block was part of the composite major adverse cardiac events that occurred in 46% of cases [63]. In fact, complete heart block is often the first indication of ICI myocarditis and can be fatal [80].

Treatment of advanced conduction disease and/or bradycardia first involves cessation of the offending agent. In the case of hemodynamic instability and/or serious symptoms, temporary transvenous pacing should be considered [115]. In many situations, recovery of conduction is possible, but if heart block persists more than 3–5 days, then a permanent pacemaker should be implanted [116]. In the case of myocarditis, there should be a low threshold to also implant an ICD if ventricular arrhythmias and left ventricular systolic dysfunction coexist with heart block [116].

Arrhythmic complications of cancer therapeutics are an emerging area of recognition and concern. In most cases, arrhythmias can be managed without significant sequelae. Rarely however, arrhythmias can present as life-threatening emergencies that require prompt medical attention. Specific cancer treatments can lead to sustained symptomatic ventricular tachvarrhythmias or advanced conduction system disease, often necessitating resuscitative efforts. In particular, various targeted therapies are associated with QT prolongation, which can lead to TdP, and certain treatments are directly arrhythmogenic to the ventricular myocardium. Advanced conduction system disease is most commonly associated with ICI myocarditis. Given the severity of these toxicities, improved screening and monitoring strategies must be developed to minimize patient risk while allowing them to continue receiving necessary and potentially lifesaving cancer treatment.

Valvular Heart Disease

Therapeutic advances in chest radiotherapy have improved survival in patients with thoracic malignancy. As a result, patients are living long enough to unveil latent cardiac toxicity [117]. Although radiation-induced heart disease (RIHD) can manifest in a variety of pathologies, valvular disease imposes a high cardiovascular burden in cancer survivors and is associated with significant therapeutic challenges [117].

Pathophysiology

RIHD is largely attributed to direct endothelial damage caused by transient increases in oxidative stress and a resultant pro-inflammatory state. This leads to the upregulation of matrix metalloproteinase, adhesion molecules, cytokines, and collagen and is the hallmark of radiation-induced fibrosis [118]. Further to this, direct deposition of collagen on the heart valves leads to leaflet thickening and ultimately calcification resulting in regurgitation and/or stenosis [118]. The aortomitral curtain is classically affected and can mimic rheumatic heart disease, although is distinguished by the commissural fissure which is characteristically lost in rheumatic heart disease but not RIHD [119]. This cellular damage may be further confounded by the atherosclerotic contribution of traditional cardiac risk factors such as hyperlipidemia and smoking [118]. Additionally, the combination of anthracycline chemotherapy and mediastinal radiation as seen in common cancers such as breast, Hodgkin's lymphoma, and non-Hodgkin's lymphoma can further amplify RIHD [120].

Management of Valvular Heart Disease

Patients presenting with an acute valvular emergency require urgent stabilization, irrespective of their malignancy status. Symptoms and signs of acute valvular emergency may include dyspnea, tachycardia, pulmonary edema, and cardiogenic shock; patients can also present with angina or syncope, particularly with exercise. Accurate diagnosis of the underlying etiology of these findings may be confounded in oncologic patients with concomitant bleeding or anemia. Additionally, treatment-related factors including tumor lysis syndrome with hemodynamically significant fluid shifts and chemotherapy-related left ventricular dysfunction can contribute to acute decompensation of otherwise stable valvular disease [121].

Radiation-induced valvular disease may manifest as any valvular abnormality; however, left-sided valves are more commonly affected, most frequently the aortic valve [122]. Symptomatic severe aortic stenosis is associated with a high mortality without intervention, irrespective of cancer status [123]. This is particularly apparent in studies from the pre-transcatheter aortic valve replacement (TAVR) era which demonstrates a significant survival advantage in oncologic patients undergoing surgical aortic valve replacement (SAVR), regardless of cancer stage [124]. Despite this, current heart team risk assessment using the Society of Thoracic Surgeons (STS) score incorporates a history of cancer and chest radiation as covariates secondary to the greater burden of periprocedural complications in this group.

Modern Techniques

The innovation of TAVR has provided a platform to treat patients at increased surgical risk. Recent studies have demonstrated favorable results in patients with a history of chest irradiation undergoing TAVR with less adjusted 30-day mortality, less postoperative atrial fibrillation, and shorter hospitalization compared with SAVR, despite older age, higher STS score, and more baseline comorbidities [125–127]. However, patient selection for TAVR in the oncologic population must involve careful consideration of disease prognosis, both from the perspective of the oncologist and the cardiologist. The outcomes of patients with active cancer undergoing TAVR in a large registry study were recently described [128]. Although effective and safe in the short term, the use of TAVR in patients with active cancer was found to carry a worse intermediate prognosis compared with patients without cancer. This was largely attributed to cancer-related death and highlights the importance of individualized assessment regarding the appropriateness or futility of invasive valvular intervention.

Cardiovascular Imaging

With rapid increase in new therapeutics that leads to improved survival in cancer patients, we observe a concomitant increase in the morbidity and mortality attributed to their side effects. While cancer remains the second most common cause of death, worldwide, after heart disease, indeed, cardiovascular disease is the most common cause of death among cancer survivors. This may result from shared risk factors by cancer and CV disease, and/or due to toxic effects of therapeutics. As we begin to recognize trends and disentangle associations of cardiovascular outcomes with certain therapies, cardiovascular oncologic programs have relied on cardiac imaging for surveillance, early recognition, and management of treatment-related cardiotoxicities in cancer patients. Advanced imaging modalities can shed light onto pathophysiologic mechanisms and often recognize sub- or preclinical manifestations.

In this section, we will describe the role and utility of various imaging modalities in the evaluation of the cardiac emergencies discussed in this chapter, including myocardial dysfunction, ischemic heart disease, valvular heart disease, pericardial disease, myocarditis, amyloid, and non-ischemic vascular dysfunction. Importantly, we address the role of multimodality imaging in assessing various components of the cardiovascular system to paint a comprehensive snapshot of a patient's cardiovascular health. Indications for serial imaging and surveillance, in accordance with current guidelines, will also be discussed in the context of the underlying neoplasia and the presumed offending chemotherapy, immunotherapy, or radiation therapy. This section focuses on echocardiography, CMR, and cardiac CT. Yet it should be noted that single-photon emission computed tomography (SPECT) and positron emission tomography (PET) are also being explored in the preclinical and clinical setting for imaging patients in cardio-oncologic, particularly for myocardial perfusion and molecular imaging to detect subclinical cardiovascular dysfunction [129].

Echocardiogram

Widespread availability of echocardiography makes it the most utilized imaging modality for assessing myocardial dysfunction. Global longitudinal strain (GLS) is a particularly valuable tool in this setting. Guideline committees recommend performing echocardiogram with a GLS speckle-tracking technique at baseline, and at clinically reasonable intervals during treatment with anthracyclines, ICIs, and TKIs. Additionally, LVEF should be assessed at least annually after the end of treatment. Similarly, echocardiography for monitoring of LVEF and strain imaging are recommended at baseline and with every three cycles (or every 3 months) of trastuzumab, VEGF inhibitors, and every 6 months in the immediate post treatment period. Importantly, in emergent situations where patients undergoing treatment present with clinical symptoms of heart failure or those with a significant drop in LVEF on routine follow-up, echocardiography is a particularly useful tool to monitor the extent of damage and response to intervention. In addition to assessment of systolic function, echocardiogram is the modality of choice for evaluation of radiation-induced valvular heart disease and diastolic dysfunction commonly associated with cardiac remodeling [130, 131].

Initial changes in GLS often precede clinical manifestations of heart failure that would require emergent action. This provides an opportunity for the clinicians to assess and discuss the benefit-risk of continuing with the current cancer treatment plan. Such discussion may result in temporary or permanent change in the treatment plan, continuation of the plan with closer monitoring, or in some cases, the addition of cardioprotective agents to alter the progression of cardiac adverse effects.

Echocardiography with strain imaging should be strongly considered when suspicion of ICI toxicity exists. A recent study has shown that GLS is reduced in these cases, independently of the LVEF being reduced or preserved [82]. The definitions of "definite," "probable," or "possible" myocarditis all involve new wall motion abnormalities on echocardiogram, combined with a clinical syndrome consistent with myocarditis, biomarker evidence of myonecrosis, EKG evidence of myopericarditis, and/or negative angiography excluding obstructive coronary disease [132]. In cases where the pericardium is involved, echocardiography is, again, the first-line modality as it can assess for effusion and provide hemodynamic data suggestive of tamponade or constrictive physiology. Stress echocardiography is valuable for the assessment of wall motion abnormality secondary to ischemic cardiomyopathy in patients undergoing radiation or endocrine therapy in patients presenting with signs of stable or unstable coronary artery disease.

Cardiac Magnetic Resonance Imaging

In recent years, the use of CMR has grown exponentially due to its accurate evaluation of chamber quantification and function. Intracellular and interstitial myocardial tissue changes are more directly visualized with T2 and T1 mapping sequences. These changes, when detected early, can provide a valuable insight to the pathophysiologic mechanisms of cancer treatment associated-myocardial dysfunction. Early detection of edema and fibrosis, measured as extracellular volume fraction (ECV), and their surveillance during treatment could inform mitigation strategies to avoid progression to a cardiac emergency. As for now, guideline committees recommend CMR instead of echocardiogram only if echo is technically unfeasible, or where echo results are discrepant or suboptimal [130, 131].

In cases of new-onset heart failure where the etiology is unclear, CMR provides volumetric and mass data that can accurately assess fibrosis burden in the setting of patients undergoing radiation therapy or to confirm noninvasively a diagnosis of cardiac amyloidosis. Similarly, it may be used for hemodynamic assessment and evaluation of pericardial inflammation. Using T2-weighted imaging (representing free water and edema), one can assess for myocardial edema, to assess for myocarditis, which is useful in acute heart failure emergencies in those receiving ICI therapy [132]. Stress perfusion by CMR is an effective and noninvasive method of quantifying myocardial perfusion reserve when an ischemic insult is suspected. This imaging modality may be considered during evaluation of suspected TKI-related ischemia.

While no specific recommendations are provided for its use in these settings, CMR is a powerful diagnostic tool that may provide more granular data than traditional echocardiograms and may serve in lieu of more invasive methods (e.g., biopsy, angiography) in the right clinical setting.

Cardiac Computed Tomography

CT scans can also play a vital role in diagnosis and prevention of cardiovascular emergencies. Cardiac CT scans with contrast can identify stenosis to assist with ischemic cardiovascular disease diagnosis or intervention decisions or calcification to aid prevention planning. While coronary gating is important for assessing stenosis, calcification can be noted on chest CTs performed during lung cancer screening or follow-up. Aortic valve calcium score can also be assessed on CT, which can be particularly helpful when considering severity in the context of patients being considered for valve procedures such as TAVR. Pericardial calcification and effusion, as well as subepicardial delayed gadolinium enhancement suggestive of myocarditis, can also be assessed on chest CT. Non-gated chest CTA is most often used to diagnose pulmonary embolism.

Cardio-Oncologic in the COVID-19 Era

The coronavirus disease 2019 (COVID-19) pandemic primarily spanning 2019–2020 posed unprecedented challenges for cardio-oncologic. The emergence of this new field of medicine coincided with a once-in-a-lifetime experience. Most clinicians and scientists will serve in at most one pandemic in their lifetimes. For some during this pandemic, their lifetimes were cut short. Healthcare workers across the globe sacrificed their lives, health, mental well-being, and time with their families in order to present daily at the frontlines and save lives. In cardio-oncologic, management decisions needed to be balanced between anticipated cardiovascular toxicities from cancer therapies and observed cardiotoxicities from COVID-19. Rapid implementation of innovative technologies related to telemedicine, artificial intelligence, and digital health paved the way for continued transformation in precision cardio-oncologic. Precision cardio-oncologic knows no bounds and has a tremendous potential to revolutionize our care for patients, using various "omics" in partnership with imaging, machine learning, computational simulation, ex vivo modeling, and mobile health [129, 133]. Combining biotechnological advances integrated in electronic health records with informatics tools for patient care in real time will likely herald a new era. The pandemic taught us how to raise the standard of emergency care, leading to a new outlook on care delivery.

References

- Shiga T, Hiraide M. Cardiotoxicities of 5-fluorouracil and other fluoropyrimidines. Curr Treat Options in Oncol. 2020;21(4):27.
- 2. Brown SA. Preventive cardio-oncology: the time has come. Front Cardiovasc Med. 2019;6:187.
- Amir E, Seruga B, Niraula S, Carlsson L, Ocana A. Toxicity of adjuvant endocrine therapy in postmenopausal breast cancer patients: a systematic review and meta-analysis. J Natl Cancer Inst. 2011;103(17):1299–309.
- Mouridsen H, Keshaviah A, Coates AS, Rabaglio M, Castiglione-Gertsch M, Sun Z, et al. Cardiovascular adverse events during adjuvant endocrine therapy for early breast cancer using letrozole or tamoxifen: safety analysis of BIG 1-98 trial. J Clin Oncol. 2007;25(36):5715–22.
- Buzdar A, Howell A, Cuzick J, Wale C, Distler W, Hoctin-Boes G, et al. Comprehensive side-effect profile of anastrozole and tamoxifen as adjuvant treatment for early-stage breast cancer: long-term safety analysis of the ATAC trial. Lancet Oncol. 2006;7(8):633–43.
- 6. Matthews A, Stanway S, Farmer RE, Strongman H, Thomas S, Lyon AR, et al. Long term adjuvant endocrine therapy and risk of

cardiovascular disease in female breast cancer survivors: systematic review. BMJ. 2018;363:k3845.

- Iliescu C, Grines CL, Herrmann J, Yang EH, Cilingiroglu M, Charitakis K, et al. SCAI expert consensus statement: evaluation, management, and special considerations of cardio-oncology patients in the cardiac catheterization laboratory (endorsed by the Cardiological Society of India, and Sociedad Latino Americana de Cardiologia Intervencionista). Catheter Cardiovasc Interv. 2016;87(5):895–9.
- Bharadwaj AS, Swamy PM, Mamas MA. Outcomes of percutaneous coronary interventions in cancer patients. Exp Ref Cardiovasc Ther. 2020;18(1):25–32.
- van Werkum JW, Heestermans AA, Zomer AC, Kelder JC, Suttorp MJ, Rensing BJ, et al. Predictors of coronary stent thrombosis: the Dutch Stent Thrombosis Registry. J Am Coll Cardiol. 2009;53(16):1399–409.
- Gross CM, Posch MG, Geier C, Olthoff H, Kramer J, Dechend R, et al. Subacute coronary stent thrombosis in cancer patients. J Am Coll Cardiol. 2008;51(12):1232–3.
- Navi BB, Reiner AS, Kamel H, Iadecola C, Okin PM, Elkind MSV, et al. Risk of arterial thromboembolism in patients with cancer. J Am Coll Cardiol. 2017;70(8):926–38.
- Ridker PM, MacFadyen JG, Thuren T, Everett BM, Libby P, Glynn RJ, et al. Effect of interleukin-1beta inhibition with canakinumab on incident lung cancer in patients with atherosclerosis: exploratory results from a randomised, double-blind, placebo-controlled trial. Lancet. 2017;390(10105):1833–42.
- Adess M, Eisner R, Nand S, Godwin J, Messmore HL, Wehrmacher WH. Thromboembolism in cancer patients: pathogenesis and treatment. Clin Appl Thromb Hemost. 2006;12(3):254–66.
- Campia U, Moslehi JJ, Amiri-Kordestani L, Barac A, Beckman JA, Chism DD, et al. Cardio-oncology: vascular and metabolic perspectives: a scientific statement from the American Heart Association. Circulation. 2019;139(13):e579–602.
- 15. Iliescu CA, Grines CL, Herrmann J, Yang EH, Cilingiroglu M, Charitakis K, et al. SCAI expert consensus statement: evaluation, management, and special considerations of cardio-oncology patients in the cardiac catheterization laboratory (endorsed by the cardiological society of India, and sociedad Latino Americana de Cardiologia intervencionista). Catheter Cardiovasc Interv. 2016;87(5):E202–23.
- Gietema JA, Sleijfer DT, Willemse PH, Schraffordt Koops H, van Ittersum E, Verschuren WM, et al. Long-term follow-up of cardiovascular risk factors in patients given chemotherapy for disseminated nonseminomatous testicular cancer. Ann Intern Med. 1992;116(9):709–15.
- 17. Bundred NJ. The effects of aromatase inhibitors on lipids and thrombosis. Br J Cancer. 2005;93(Suppl 1):S23–7.
- Lin C, Chen LS, Kuo SJ, Chen DR. Adjuvant tamoxifen influences the lipid profile in breast cancer patients. Breast Care (Basel). 2014;9(1):35–9.
- Khosrow-Khavar F, Bouganim N, Filion KB, Suissa S, Azoulay L. Cardiotoxicity of use of sequential aromatase inhibitors in women with breast Cancer. Am J Epidemiol. 2020;189(10):1086–95.
- Arnett DK, Blumenthal RS, Albert MA, Buroker AB, Goldberger ZD, Hahn EJ, et al. 2019 ACC/AHA guideline on the primary prevention of cardiovascular disease: a report of the American College of Cardiology/American Heart Association Task Force on Clinical Practice Guidelines. Circulation. 2019;140(11):e596–646.
- FitzGerald TJ, Bishop-Jodoin M, Laurie F, Lukez A, O'Loughlin L, Sacher A. Treatment toxicity: radiation. Hematol Oncol Clin North Am. 2019;33(6):1027–39.
- Yusuf SW, Sami S, Daher IN. Radiation-induced heart disease: a clinical update. Cardiol Res Pract. 2011;2011:317659.

- 23. Stewart FA, Heeneman S, Te Poele J, Kruse J, Russell NS, Gijbels M, et al. Ionizing radiation accelerates the development of atherosclerotic lesions in ApoE-/- mice and predisposes to an inflammatory plaque phenotype prone to hemorrhage. Am J Pathol. 2006;168(2):649–58.
- Madan R, Benson R, Sharma DN, Julka PK, Rath GK. Radiation induced heart disease: pathogenesis, management and review literature. J Egypt Natl Canc Inst. 2015;27(4):187–93.
- Desai MY, Windecker S, Lancellotti P, Bax JJ, Griffin BP, Cahlon O, et al. Prevention, diagnosis, and Management of Radiation-Associated Cardiac Disease: JACC scientific expert panel. J Am Coll Cardiol. 2019;74(7):905–27.
- Desai MY, Jellis CL, Kotecha R, Johnston DR, Griffin BP. Radiation-associated cardiac disease: a practical approach to diagnosis and management. JACC Cardiovasc Imaging. 2018;11(8):1132–49.
- 27. Tukenova M, Guibout C, Oberlin O, Doyon F, Mousannif A, Haddy N, Guérin S, Pacquement H, Aouba A, Hawkins M, Winter D, Bourhis J, Lefkopoulos D, Diallo I, de Vathaire F. Long-term overall and cardiovascular mortality following childhood cancer: the role of cancer treatment. Radioprotection. 2008;43:1308–15.
- Galper SL, Yu JB, Mauch PM, Strasser JF, Silver B, Lacasce A, et al. Clinically significant cardiac disease in patients with Hodgkin lymphoma treated with mediastinal irradiation. Blood. 2011;117(2):412–8.
- Clasen SC, Ky B, O'Quinn R, Giantonio B, Teitelbaum U, Carver JR. Fluoropyrimidine-induced cardiac toxicity: challenging the current paradigm. J Gastrointest Oncol. 2017;8(6):970–9.
- Hu JR, Florido R, Lipson EJ, Naidoo J, Ardehali R, Tocchetti CG, et al. Cardiovascular toxicities associated with immune checkpoint inhibitors. Cardiovasc Res. 2019;115(5):854–68.
- Herrmann J, Yang EH, Iliescu CA, Cilingiroglu M, Charitakis K, Hakeem A, et al. Vascular toxicities of cancer therapies: the old and the new–an evolving avenue. Circulation. 2016;133(13):1272–89.
- 32. Latifi Y, Moccetti F, Wu M, Xie A, Packwood W, Qi Y, et al. Thrombotic microangiopathy as a cause of cardiovascular toxicity from the BCR-ABL1 tyrosine kinase inhibitor ponatinib. Blood. 2019;133(14):1597–606.
- Piccirillo JF, Tierney RM, Costas I, Grove L, Spitznagel EL. Prognostic importance of comorbidity in a hospital-based cancer registry. JAMA. 2004;291(20):2441–7.
- 34. Small HY, Montezano AC, Rios FJ, Savoia C, Touyz RM. Hypertension due to antiangiogenic cancer therapy with vascular endothelial growth factor inhibitors: understanding and managing a new syndrome. Can J Cardiol. 2014;30(5):534–43.
- 35. de Jesus-Gonzalez N, Robinson E, Penchev R, von Mehren M, Heinrich MC, Tap W, et al. Regorafenib induces rapid and reversible changes in plasma nitric oxide and endothelin-1. Am J Hypertens. 2012;25(10):1118–23.
- 36. Robinson ES, Khankin EV, Choueiri TK, Dhawan MS, Rogers MJ, Karumanchi SA, et al. Suppression of the nitric oxide pathway in metastatic renal cell carcinoma patients receiving vascular endothelial growth factor-signaling inhibitors. Hypertension. 2010;56(6):1131–6.
- Lankhorst S, Saleh L, Danser AJ, van den Meiracker AH. Etiology of angiogenesis inhibition-related hypertension. Curr Opin Pharmacol. 2015;21:7–13.
- Yusuf SW, Razeghi P, Yeh ET. The diagnosis and management of cardiovascular disease in cancer patients. Curr Probl Cardiol. 2008;33(4):163–96.
- Razak N, Jones G, Bhandari M, Berndt M, Metharom P. Cancerassociated thrombosis: an overview of mechanisms, risk factors, and treatment. Cancers (Basel). 2018;10(10):380.
- Caine GJ, Stonelake PS, Lip GY, Kehoe ST. The hypercoagulable state of malignancy: pathogenesis and current debate. Neoplasia. 2002;4(6):465–73.

- 41. Anders J, Grigsby P, Singh A. Cisplatin chemotherapy (without erythropoietin) and risk of life-threatening thromboembolic events in carcinoma of the uterine cervix: the tip of the iceberg? A review of the literature. Radiat Oncol. 2006;1. https://doi. org/10.1186/1748-717X-1-14.
- 42. Rodeghiero F, Elice F. Thalidomide and thrombosis. Pathophysiol Haemost Thromb. 2003;33(Suppl 1):15–8.
- 43. Kuenen BC, Levi M, Meijers JC, Kakkar AK, van Hinsbergh VW, Kostense PJ, et al. Analysis of coagulation cascade and endothelial cell activation during inhibition of vascular endothelial growth factor/vascular endothelial growth factor receptor pathway in cancer patients. Arterioscler Thromb Vasc Biol. 2002;22(9):1500–5.
- 44. Khorana AA, Kuderer NM, Culakova E, Lyman GH, Francis CW. Development and validation of a predictive model for chemotherapy-associated thrombosis. Blood. 2008;111(10):4902–7.
- 45. Key NS, Khorana AA, Kuderer NM, Bohlke K, Lee AYY, Arcelus JI, et al. Venous thromboembolism prophylaxis and treatment in patients with cancer: ASCO clinical practice guideline update. J Clin Oncol. 2020;38(5):496–520.
- 46. Fesmire FM, Brown MD, Espinosa JA, Shih RD, Silvers SM, Wolf SJ, et al. Critical issues in the evaluation and management of adult patients presenting to the emergency department with suspected pulmonary embolism. Ann Emerg Med. 2011;57(6):628– 52.e75.
- 47. Mody P, Bikdeli B, Wang Y, Imazio M, Krumholz HM. Trends in acute pericarditis hospitalizations and outcomes among the elderly in the USA, 1999–2012. Eur Heart J Qual Care Clin Outcomes. 2018;4(2):98–105.
- Imazio M, Cecchi E, Demichelis B, Ierna S, Demarie D, Ghisio A, et al. Indicators of poor prognosis of acute pericarditis. Circulation. 2007;115(21):2739–44.
- Ghosh AK, Crake T, Manisty C, Westwood M. Pericardial disease in cancer patients. Curr Treat Options Cardiovasc Med. 2018;20(7):60.
- 50. Klein AL, Abbara S, Agler DA, Appleton CP, Asher CR, Hoit B, et al. American Society of Echocardiography clinical recommendations for multimodality cardiovascular imaging of patients with pericardial disease: endorsed by the Society for Cardiovascular Magnetic Resonance and Society of Cardiovascular Computed Tomography. J Am Soc Echocardiogr. 2013;26(9):965–1012.e15.
- Quint LE. Thoracic complications and emergencies in oncologic patients. Cancer Imaging. 2009;9(Spec No A):S75–S82.
- 52. Chen DY, Huang WK, Chien-Chia Wu V, Chang WC, Chen JS, Chuang CK, et al. Cardiovascular toxicity of immune checkpoint inhibitors in cancer patients: a review when cardiology meets immuno-oncology. J Formos Med Assoc. 2019;119:1461–75.
- Jeon HW, Cho DG, Park JK, Hyun KY, Choi SY, Suh JH, et al. Prognostic factors affecting survival of patients with cancerrelated pericardial effusion managed by surgery. World J Surg Oncol. 2014;12:249.
- 54. Chang HM, Okwuosa TM, Scarabelli T, Moudgil R, Yeh ETH. Cardiovascular complications of cancer therapy: best practices in diagnosis, prevention, and management: part 2. J Am Coll Cardiol. 2017;70(20):2552–65.
- Dogan SE, Mizrak D, Alkan A, Demirkazik A. Docetaxel-induced pericardial effusion. J Oncol Pharm Pract. 2017;23(5):389–91.
- Curigliano G, Lenihan D, Fradley M, Ganatra S, Barac A, Blaes A, et al. Management of cardiac disease in cancer patients throughout oncological treatment: ESMO consensus recommendations. Ann Oncol. 2020;31(2):171–90.
- Avila MS, Siqueira SRR, Ferreira SMA, Bocchi EA. Prevention and treatment of chemotherapy-induced cardiotoxicity. Methodist Debakey Cardiovasc J. 2019;15(4):267–73.
- Babiker HM, McBride A, Newton M, Boehmer LM, Drucker AG, Gowan M, et al. Cardiotoxic effects of chemotherapy: a review

of both cytotoxic and molecular targeted oncology therapies and their effect on the cardiovascular system. Crit Rev Oncol Hematol. 2018;126:186–200.

- 59. Camara Planek MI, Silver AJ, Volgman AS, Okwuosa TM. Exploratory review of the role of statins, colchicine, and aspirin for the prevention of radiation-associated cardiovascular disease and mortality. J Am Heart Assoc. 2020;9(2):e014668.
- Han X, Zhou Y, Liu W. Precision cardio-oncology: understanding the cardiotoxicity of cancer therapy. NPJ Precis Oncol. 2017;1(1):31.
- Keramida K, Parissis JT, Chioncel O, Farmakis D. Cardiogenic shock in cancer. Heart Fail Rev. 2019;24(6):997–1004.
- 62. Laufer-Perl M, Arnold JH, Mor L, Amrami N, Derakhshesh M, Moshkovits Y, et al. The association of reduced global longitudinal strain with cancer therapy-related cardiac dysfunction among patients receiving cancer therapy. Clin Res Cardiol. 2020;109(2):255–62.
- Mahmood SS, Fradley MG, Cohen JV, Nohria A, Reynolds KL, Heinzerling LM, et al. Myocarditis in patients treated with immune checkpoint inhibitors. J Am Coll Cardiol. 2018;71(16):1755–64.
- 64. Dobbin SJH, Cameron AC, Petrie MC, Jones RJ, Touyz RM, Lang NN. Toxicity of cancer therapy: what the cardiologist needs to know about angiogenesis inhibitors. Heart. 2018;104(24):1995–2002.
- Okazaki T, Tanaka Y, Nishio R, Mitsuiye T, Mizoguchi A, Wang J, et al. Autoantibodies against cardiac troponin I are responsible for dilated cardiomyopathy in PD-1-deficient mice. Nat Med. 2003;9(12):1477–83.
- 66. Dolladille C, Ederhy S, Allouche S, Dupas Q, Gervais R, Madelaine J, et al. Late cardiac adverse events in patients with cancer treated with immune checkpoint inhibitors. J Immunother Cancer. 2020;8(1):e000261.
- Heidenreich PA, Hancock SL, Vagelos RH, Lee BK, Schnittger I. Diastolic dysfunction after mediastinal irradiation. Am Heart J. 2005;150(5):977–82.
- Hancock SL, Tucker MA, Hoppe RT. Factors affecting late mortality from heart disease after treatment of Hodgkin's disease. JAMA. 1993;270(16):1949–55.
- 69. Upshaw JN, Finkelman B, Hubbard RA, Smith AM, Narayan HK, Arndt L, et al. Comprehensive assessment of changes in left ventricular diastolic function with contemporary breast cancer therapy. JACC Cardiovasc Imaging. 2020;13(1 Pt 2):198–210.
- Demissei BG, Hubbard RA, Zhang L, Smith AM, Sheline K, McDonald C, et al. Changes in cardiovascular biomarkers with breast cancer therapy and associations with cardiac dysfunction. J Am Heart Assoc. 2020;9(2):e014708.
- Michel L, Rassaf T, Totzeck M. Biomarkers for the detection of apparent and subclinical cancer therapy-related cardiotoxicity. J Thorac Dis. 2018;10(Suppl 35):S4282–S95.
- Totzeck M, Schuler M, Stuschke M, Heusch G, Rassaf T. Cardiooncology - strategies for management of cancer-therapy related cardiovascular disease. Int J Cardiol. 2019;280:163–75.
- Hryniewicki AT, Wang C, Shatsky RA, Coyne CJ. Management of immune checkpoint inhibitor toxicities: a review and clinical guideline for emergency physicians. J Emerg Med. 2018;55(4):489–502.
- 74. Brahmer JR, Lacchetti C, Schneider BJ, Atkins MB, Brassil KJ, Caterino JM, et al. Management of Immune-Related Adverse Events in patients treated with immune checkpoint inhibitor therapy: American Society of Clinical Oncology clinical practice guideline. J Clin Oncol. 2018;36(17):1714–68.
- Jain V, Mohebtash M, Rodrigo ME, Ruiz G, Atkins MB, Barac A. Autoimmune myocarditis caused by immune checkpoint inhibitors treated with antithymocyte globulin. J Immunother. 2018;41(7):332–5.

- Esfahani K, Buhlaiga N, Thebault P, Lapointe R, Johnson NA, Miller WH Jr. Alemtuzumab for immune-related myocarditis due to PD-1 therapy. N Engl J Med. 2019;380(24):2375–6.
- Postow MA, Sidlow R, Hellmann MD. Immune-related adverse events associated with immune checkpoint blockade. N Engl J Med. 2018;378(2):158–68.
- Zhang L, Awadalla M, Mahmood SS, Nohria A, Hassan MZO, Thuny F, et al. Cardiovascular magnetic resonance in immune checkpoint inhibitor-associated myocarditis. Eur Heart J. 2020;41(18):1733–43.
- Salem J-E, Manouchehri A, Moey M, Lebrun-Vignes B, Bastarache L, Pariente A, et al. Cardiovascular toxicities associated with immune checkpoint inhibitors: an observational, retrospective, pharmacovigilance study. Lancet Oncol. 2018;19(12):1579–89.
- Johnson DB, Balko JM, Compton ML, Chalkias S, Gorham J, Xu Y, et al. Fulminant myocarditis with combination immune checkpoint blockade. N Engl J Med. 2016;375(18):1749–55.
- Moslehi JJ, Salem J-E, Sosman JA, Lebrun-Vignes B, Johnson DB. Increased reporting of fatal immune checkpoint inhibitorassociated myocarditis. Lancet. 2018;391(10124):933.
- Awadalla M, Mahmood SS, Groarke JD, Hassan MZO, Nohria A, Rokicki A, et al. Global longitudinal strain and cardiac events in patients with immune checkpoint inhibitor-related myocarditis. J Am Coll Cardiol. 2020;75(5):467–78.
- Sobol I, Chen CL, Mahmood SS, Borczuk AC. Histopathologic characterization of myocarditis associated with immune checkpoint inhibitor therapy. Arch Pathol Lab Med. 2020; https://doi. org/10.5858/arpa.2019-0447-OA. Epub ahead of print
- 84. Zhang L, Zlotoff DA, Awadalla M, Mahmood SS, Nohria A, Hassan MZO, et al. Major adverse cardiovascular events and the timing and dose of corticosteroids in immune checkpoint inhibitor-associated myocarditis. Circulation. 2020;141(24):2031–4.
- Mahmood SS, Chen CL, Shapnik N, Krishnan U, Singh HS, Makker V. Myocarditis with tremelimumab plus durvalumab combination therapy for endometrial cancer: a case report. Gynecol Oncol Rep. 2018;25:74–7.
- Salem J-E, Allenbach Y, Vozy A, Brechot N, Johnson DB, Moslehi JJ, et al. Abatacept for severe immune checkpoint inhibitor-associated myocarditis. N Engl J Med. 2019;380(24):2377–9.
- Esfahani K, Buhlaiga N, Thébault P, Lapointe R, Johnson NA, Miller WH. Alemtuzumab for immune-related myocarditis due to PD-1 therapy. N Engl J Med. 2019;380(24):2375–6.
- Rhea I, Burgos PH, Fradley MG. Arrhythmogenic anticancer drugs in cardio-oncology. Cardiol Clin. 2019;37(4):459–68.
- 89. January CT, Wann LS, Alpert JS, Calkins H, Cigarroa JE, Cleveland JC Jr, et al. 2014 AHA/ACC/HRS guideline for the management of patients with atrial fibrillation: a report of the American College of Cardiology/American Heart Association task force on practice guidelines and the Heart Rhythm Society. Circulation. 2014;130(23):e199–267.
- Al-Khatib SM, LaPointe NM, Kramer JM, Califf RM. What clinicians should know about the QT interval. JAMA. 2003;289(16):2120–7.
- 91. Drew BJ, Ackerman MJ, Funk M, Gibler WB, Kligfield P, Menon V, et al. Prevention of torsade de pointes in hospital settings: a scientific statement from the American Heart Association and the American College of Cardiology Foundation. Circulation. 2010;121(8):1047–60.
- Chandrasekhar S, Fradley MG. QT interval prolongation associated with cytotoxic and targeted cancer therapeutics. Curr Treat Options in Oncol. 2019;20(7):55.
- Naing A, Veasey-Rodrigues H, Hong DS, Fu S, Falchook GS, Wheler JJ, et al. Electrocardiograms (ECGs) in phase I anticancer

drug development: the MD Anderson Cancer Center experience with 8518 ECGs. Ann Oncol. 2012;23(11):2960–3.

- 94. Roboz GJ, Ritchie EK, Carlin RF, Samuel M, Gale L, Provenzano-Gober JL, et al. Prevalence, management, and clinical consequences of QT interval prolongation during treatment with arsenic trioxide. J Clin Oncol. 2014;32(33):3723–8.
- 95. U.S. Food and Drug Administration. Arsenic trioxide. Revised Feb 2015.Cephalon/Teva Pharmaceuticals USA, North Wales PA. https://www.accessdata.fda.gov/drugsatfda_docs/ label/2015/021248s013lbl.pdf.
- 96. Bello CL, Mulay M, Huang X, Patyna S, Dinolfo M, Levine S, et al. Electrocardiographic characterization of the QTc interval in patients with advanced solid tumors: pharmacokineticpharmacodynamic evaluation of sunitinib. Clin Cancer Res. 2009;15(22):7045–52.
- Kim TD, le Coutre P, Schwarz M, Grille P, Levitin M, Fateh-Moghadam S, et al. Clinical cardiac safety profile of nilotinib. Haematologica. 2012;97(6):883–9.
- U.S. Food and Drug Administration. Nilotinib. Revised Jun 2010. Novartis, East Hanover NJ. https://www.accessdata.fda.gov/drugsatfda_docs/label/2010/022068s004s005lbl.pdf.
- 99. Zang J, Wu S, Tang L, Xu X, Bai J, Ding C, et al. Incidence and risk of QTc interval prolongation among cancer patients treated with vandetanib: a systematic review and meta-analysis. PLoS One. 2012;7(2):e30353.
- 100. Porta-Sanchez A, Gilbert C, Spears D, Amir E, Chan J, Nanthakumar K, et al. Incidence, diagnosis, and management of QT prolongation induced by cancer therapies: a systematic review. J Am Heart Assoc. 2017;6(12):e007724.
- 101. Hortobagyi GN, Stemmer SM, Burris HA, Yap YS, Sonke GS, Paluch-Shimon S, et al. Updated results from MONALEESA-2, a phase III trial of first-line ribociclib plus letrozole versus placebo plus letrozole in hormone receptor-positive, HER2-negative advanced breast cancer. Ann Oncol. 2018;29(7):1541–7.
- U.S. Food and Drug Administration. Ribociclib. Issued Mar 2017. Novartis, East Hanover NJ.https://www.accessdata.fda.gov/drugsatfda_docs/label/2017/209092s000lbl.pdf.
- 103. Fradley MG, Gliksman M, Emole J, Viganego F, Rhea I, Welter-Frost A, et al. Rates and risk of atrial arrhythmias in patients treated with ibrutinib compared with cytotoxic chemotherapy. Am J Cardiol. 2019;124(4):539–44.
- 104. Guha A, Derbala MH, Zhao Q, Wiczer TE, Woyach JA, Byrd JC, et al. Ventricular arrhythmias following ibrutinib initiation for lymphoid malignancies. J Am Coll Cardiol. 2018;72(6):697–8.
- 105. Salem JE, Manouchehri A, Bretagne M, Lebrun-Vignes B, Groarke JD, Johnson DB, et al. Cardiovascular toxicities associated with ibrutinib. J Am Coll Cardiol. 2019;74(13):1667–78.
- 106. Chitturi K, Xu J, Araujo-Gutierrez R, Bhimaraj A, Guha A, Hussain I, et al. Immune checkpoint inhibitor-related adverse cardiovascular events in patients with lung cancer. J Am Coll Cardiol CardioOnc. 2019;1(2):182–92.
- 107. Escudier M, Cautela J, Malissen N, Ancedy Y, Orabona M, Pinto J, et al. Clinical features, management, and outcomes of immune checkpoint inhibitor-related cardiotoxicity. Circulation. 2017;136(21):2085–7.
- Mazur M, Wang F, Hodge DO, Siontis BL, Beinborn DS, Villarraga HR, et al. Burden of cardiac arrhythmias in patients with anthracycline-related cardiomyopathy. JACC Clin Electrophysiol. 2017;3(2):139–50.
- 109. Fradley MG, Viganego F, Kip K, Martin A, Patel AA, Ismail-Khan R, et al. Rates and risk of arrhythmias in cancer survivors with chemotherapy-induced cardiomyopathy compared with patients with other cardiomyopathies. Open Heart. 2017;4(2):e000701.
- 110. Hrovatin E, Viel E, Lestuzzi C, Tartuferi L, Zardo F, Brieda M, et al. Severe ventricular dysrhythmias and silent ischemia dur-

ing infusion of the antimetabolite 5-fluorouracil and cis-platin. J Cardiovasc Med (Hagerstown). 2006;7(8):637–40.

- 111. Al-Khatib SM, Stevenson WG, Ackerman MJ, Bryant WJ, Callans DJ, Curtis AB, et al. 2017 AHA/ACC/HRS guideline for Management of Patients with Ventricular Arrhythmias and the prevention of sudden cardiac death: executive summary: a report of the American College of Cardiology/American Heart Association task force on clinical practice guidelines and the Heart Rhythm Society. J Am Coll Cardiol. 2018;72(14):1677–749.
- 112. Singh JP, Solomon SD, Fradley MG, Barac A, Kremer KA, Beck CA, et al. Association of cardiac resynchronization therapy with change in left ventricular ejection fraction in patients with chemotherapy-induced cardiomyopathy. JAMA. 2019;322(18):1799–805.
- 113. Ou SH, Tang Y, Polli A, Wilner KD, Schnell P. Factors associated with sinus bradycardia during crizotinib treatment: a retrospective analysis of two large-scale multinational trials (PROFILE 1005 and 1007). Cancer Med. 2016;5(4):617–22.
- 114. Arbuck SG, Strauss H, Rowinsky E, Christian M, Suffness M, Adams J, et al. A reassessment of cardiac toxicity associated with Taxol. J Natl Cancer Inst Monogr. 1993;15:117–30.
- 115. Kusumoto FM, Schoenfeld MH, Barrett C, Edgerton JR, Ellenbogen KA, Gold MR, et al. 2018 ACC/AHA/HRS guideline on the evaluation and Management of Patients with Bradycardia and Cardiac Conduction Delay: a report of the American College of Cardiology/American Heart Association task force on clinical practice guidelines and the Heart Rhythm Society. Circulation. 2019;140(8):e382–482.
- Ali-Ahmed F, Dalgaard F, Al-Khatib SM. Sudden cardiac death in patients with myocarditis: evaluation, risk stratification, and management. Am Heart J. 2020;220:29–40.
- 117. Jaworski C, Mariani JA, Wheeler G, Kaye DM. Cardiac complications of thoracic irradiation. J Am Coll Cardiol. 2013;61(23):2319–28.
- 118. Tribble DL, Barcellos-Hoff MH, Chu BM, Gong EL. Ionizing radiation accelerates aortic lesion formation in fat-fed mice via SOD-inhibitable processes. Arterioscler Thromb Vasc Biol. 1999;19(6):1387–92.
- 119. Krapf L, Dreyfus J, Cueff C, Lepage L, Brochet E, Vahanian A, et al. Anatomical features of rheumatic and non-rheumatic mitral stenosis: potential additional value of three-dimensional echocardiography. Arch Cardiovasc Dis. 2013;106(2):111–5.
- 120. Aleman BM, van den Belt-Dusebout AW, De Bruin ML, van't Veer MB, Baaijens MH, de Boer JP, et al. Late cardiotoxicity after treatment for Hodgkin lymphoma. Blood. 2007;109(5):1878–86.
- 121. Weeks AC, Kimple ME. Spontaneous tumor lysis syndrome: a case report and critical evaluation of current diagnostic criteria and optimal treatment regimens. J Investig Med High Impact Case Rep. 2015;3(3):2324709615603199.
- 122. Donnellan E, Phelan D, McCarthy CP, Collier P, Desai M, Griffin B. Radiation-induced heart disease: a practical guide to diagnosis and management. Cleve Clin J Med. 2016;83(12):914–22.
- 123. Malaisrie SC, McDonald E, Kruse J, Li Z, McGee EC, Abicht TO, et al. Mortality while waiting for aortic valve replacement. Ann Thorac Surg. 2014;98(5):1564–70. discussion 70-1
- 124. Yusuf SW, Sarfaraz A, Durand JB, Swafford J, Daher IN. Management and outcomes of severe aortic stenosis in cancer patients. Am Heart J. 2011;161(6):1125–32.
- 125. Zhang D, Guo W, Al-Hijji MA, El Sabbagh A, Lewis BR, Greason K, et al. Outcomes of patients with severe symptomatic aortic valve stenosis after chest radiation: transcatheter versus surgical aortic valve replacement. J Am Heart Assoc. 2019;8(10):e012110.

- 126. Murphy A, Koshy A, Cameron W, Horrigan M, Yeo B, Farouque O, et al. TCT-708 outcomes of transcatheter aortic valve replacement in oncology patients with severe aortic stenosis. J Am Coll Cardiol. 2019;74(13 Suppl):B695.
- 127. Guha A, Dey AK, Arora S, Cavender MA, Vavalle JP, Sabik JF III, et al. Contemporary trends and outcomes of percutaneous and surgical aortic valve replacement in patients with cancer. J Am Heart Assoc. 2020;9(2):e014248.
- Landes U, Iakobishvili Z, Vronsky D, Zusman O, Barsheshet A, Jaffe R, et al. Transcatheter aortic valve replacement in oncology patients with severe aortic stenosis. JACC Cardiovasc Interv. 2019;12(1):78–86.
- Dreyfuss AD, Bravo PE, Koumenis C, Ky B. Precision cardiooncology. J Nucl Med. 2019;60(4):443–50.

- 130. Biersmith MA, Tong MS, Guha A, Simonetti OP, Addison D. Multimodality cardiac imaging in the era of emerging cancer therapies. J Am Heart Assoc. 2020;9(2):e013755.
- 131. Plana JC, Thavendiranathan P, Bucciarelli-Ducci C, Lancellotti P. Multi-modality imaging in the assessment of cardiovascular toxicity in the cancer patient. JACC Cardiovasc Imaging. 2018;11(8):1173–86.
- 132. Bonaca MP, Olenchock BA, Salem JE, Wiviott SD, Ederhy S, Cohen A, et al. Myocarditis in the setting of cancer therapeutics: proposed case definitions for emerging clinical syndromes in cardio-oncology. Circulation. 2019;140(2):80–91.
- 133. Brown SA, Ray JC, Herrmann J. Precision cardio-oncology: a systems-based perspective on cardiotoxicity of tyrosine kinase inhibitors and immune checkpoint inhibitors. J Cardiovasc Transl Res. 2020;13(3):402–16.

Superior Vena Cava Syndrome

Thomas C. Williamson, David E. Manthey, and Leslie R. Ellis

Case Study

A 65-year-old man with a 40-pack year smoking history presents to the emergency department complaining of shortness of breath and facial swelling that has acutely worsened over the last 24 hours. His vital signs are significant for O_2 saturation of 92% on 3 liters nasal cannula, respirations of 14/minute, blood pressure of 132/60, and heart rate of 100. Physical exam is significant for facial plethora, temporal wasting, decreased breath sounds over the right upper lung field, 1.5 cm left supraclavicular lymph node, ascites, and bilateral lower extremity edema. CT angiogram is significant for near complete occlusion of the superior vena cava without collateralization, a right upper lobe mass, and mediastinal and supraclavicular lymphadenopathy. The patient is admitted to the hospital and a bedside fine needle aspirate of the supraclavicular node is consistent with small cell carcinoma of lung origin. Further staging imaging, including MRI brain, does not reveal metastases. Treatment with chemotherapy (carboplatin/etoposide) is initiated.

Etiology/Epidemiology

Superior vena cava syndrome (SVCS) was first described by the Scottish physician William Hunter in 1757 in a case involving syphilitic infection of the aorta [1]. SVCS affects at least 15,000 people in the United States annually, occurring most commonly in patients between the ages 50 and 70 years [2]. Historically, the etiology of SVCS was predominantly infectious (syphilis, tuberculosis); however, at present, the majority of cases are due to malignancy. A review of 78 patients with SVC syndrome identified malignancy as the etiology in 60% of cases [3]. When malignancy is the cause, it is most commonly associated with non-small-cell lung cancer (50%), small-cell lung cancer (25%), lymphoma (10%), and metastatic lesions (10%), typically from breast cancer (Fig. 22.1). SVCS is actually found in a greater percentage of patients who have small-cell lung cancer (10%) versus non-small-cell lung cancer (less than 2%), but because of the greater incidence of non-small-cell lung cancer, the majority of cases of SVCS seen will be in this population [4]. Lymphomas associated with SVCS are overwhelmingly non-Hodgkin lymphomas, despite the fact that Hodgkin lymphoma typically presents with mediastinal lymphadenopathy. Common subtypes of non-Hodgkin lymphoma associated with SVCS include the diffuse large B-cell and lymphoblastic lymphoma subtypes. A type of lymphoma called primary mediastinal large B-cell lymphoma with sclerosis, although rarer, has a greater likelihood of being associated with SVCS when present, with up to 57% of patients having SVCS as presentation and over 80% having some radiographic evidence for the potential for SVCS [5, 6]. Obstruction of the superior vena cava in these situations is typically due to external compression from either the primary mass or an involved lymph node, although tumors less likely to be associated with SVCS can cause SVCS via intravascular extension and occlusion, such as intravascular large cell lymphoma or thymoma, among others [7, 8]. Approximately 2% of all cancer patients will develop some degree of SVCS [2, 9].

Surprisingly, up to 40% of the cases are now caused by indwelling lines secondary to an intrinsic thrombus associated with the line. In nonmalignant SVCS, it is estimated that intravascular devices account for up to 70% of cases [3]. This is felt to be due to not only compromised intraluminal laminar flow due to the presence of the indwelling line but also from alteration of the integrity of the vessel wall due to irritation of the endothelium by the catheter tip or wire lead [10]. However, only 1-3% of patients with central venous

© Springer Nature Switzerland AG 2021



T. C. Williamson \cdot L. R. Ellis (\boxtimes)

Division of Hematology and Oncology, Department of Internal Medicine, Comprehensive Cancer Center, Wake Forest Baptist Health, Winston-Salem, NC, USA e-mail: Irellis@wakehealth.edu

D. E. Manthey Department of Emergency Medicine, Wake Forest Baptist Health, Winston-Salem, NC, USA

K. H. Todd et al. (eds.), Oncologic Emergency Medicine, https://doi.org/10.1007/978-3-030-67123-5_22

Malignancies Causing SVC Syndrome



Fig. 22.1 Distribution of malignancies causing superior vena cava syndrome. (From McCurdy and Shanholtz [16], with permission from Wolters Kluwer Health and the Society of Critical Care Medicine)

catheters will become symptomatic [11]. This is commonly referred to as "benign" SVCS as it is not due to malignancy or alternatively because it is caused by intrinsic clot rather than external compression. A patient with benign SVCS is more often in the 30-40-year age range. Approximately 25% of pacemaker insertions are associated with obstruction or stenosis of the central venous system. It is reportedly less in patients with implanted pacemakers at an incidence of 0.2-3.3% [12]. Other nonmalignant causes of SVCS include mediastinal fibrosis, vascular diseases (atherosclerotic), infection (histoplasmosis, tuberculosis, syphilis, and actinomycosis), goiter, benign mediastinal tumors (cystic hygroma, thymoma, and teratoma), pericardial constriction, and nephritic syndrome. In pediatric patients, the cause of SVCS is most commonly iatrogenic secondary to indwelling central lines, ventriculoperitoneal shunts, or due to complications of cardiovascular surgery rather than caused by a cancer.

Anatomy

To fully understand the signs and symptoms of SVCS, the corresponding anatomy should be reviewed. The superior vena cava is 4–6 cm in length and 1.5–2 cm in width in an adult. It originates from the left and right innominate veins

and terminates into the right atrium. The SVC does not contain any venous valves. The vein itself is thin walled and therefore easily compressible. It is surrounded by lymph nodes that may become enlarged and cause external compression. It is joined just above the right atrium by the azygos vein, which drains the veins from posterior aspect of the abdominal to chest cavities. The azygos vein can become an alternative drainage system for the upper extremity through collateral veins if it is not obstructed (Fig. 22.2). An obstruction of the SVC above the azygos vein, and therefore not involving the azygos vein, may present with less severe and more insidious onset of symptoms. Other potential sites for collateral development include the internal mammary veins, lateral thoracic veins, paraspinal veins, and esophageal veins, although the ability of these systems to fully compensate for SVC obstruction is more limited than the azygos vein system. The development of these collaterals usually occurs over a period of weeks; therefore, SVCS due to rapid tumor growth may not be associated with sufficient collateral development and symptoms may be more severe. When the vena cava is obstructed, the venous pressure in the cervical veins increases to 20-40 mmHg, significantly higher than the normal range of 2–10 mmHg [13]. Development of collaterals and the rapidity of the obstruction will affect the actual values.

Fig. 22.2 Anatomy of the superior vena cava and veins of the mediastinum. (From Drews and Rabkin [14], with permission. Accessed on 27 July 2020. Copyright © 2020 UpToDate, Inc. For more information visit www. uptodate.com)



Clinical Features (Signs/Symptoms)

The clinical features of SVCS align with what you would expect to encounter due to obstruction of the major venous drainage of the head, neck, and upper extremities (Fig. 22.3 panel a). The most common sign of SVCS is facial edema (82%), which may be most noticeable in the morning after being in a recumbent position or with bending forward. Even in the presence of marked physical changes due to facial edema, this is rarely of clinical consequence [15]. Additional signs include distended neck veins (63%), distended chest veins (53%), arm edema (46%), and facial plethora, also referred to as a red ruddy complexion (20%) [16]. An interesting sign is the Pemberton sign, which is the exaggeration of the edema and flushing of the face due to the placement of the patient's arms over the head [17].

Most patients have symptoms for 2–4 weeks before a diagnosis is made. As discussed earlier, the development of collateral vessels may delay the onset of symptoms and

signs. The most common symptoms of SVCS are cough (54%) and dyspnea (54%). The latter is commonly due to an associated pleural effusion and is seen more often in malignancy-associated SVCS. Additional symptoms include hoarseness (17%), syncope (10%), headaches (9%), and dizziness (6%). Stridor may occur in 4% of patients due to swelling around the trachea. Confusion (4%) and obtundation (2%) may also be present if cerebral edema is occurring but are unlikely to be the only complaint. Visual symptoms (2%) are also rarely encountered. It is important to note that improvement in these symptoms does not necessarily represent resolution of SVCS but instead may reflect the development of an adequate venous collateral system. Additionally, symptoms may not be due solely to SVCS but due to direct compression of additional structures by the primary or metastatic mass. This pearl is especially important when considering altered mental status or neurological findings, which may be due to increased venous pressure in the cranial vault or due to brain metastases from the primary tumor [18]. The



Fig. 22.3 Clinical findings in a patient with superior vena cava syndrome, including facial edema, plethora, jugular venous distention, and prominent superficial vascularity of the upper chest and neck, are shown in panel (**a**). The vascular anatomy of the upper chest, including

the heart, superior vena cava, inferior vena cava, and subclavian vessels, is shown in panel (b). The tumor is shown compressing the superior vena cava. (From Wilson et al. [2]. ©2007 Massachusetts Medical Society, with permission)

most concerning signs and symptoms are those suggesting respiratory compromise or cerebral edema, as this may be life threatening. Because over 70% of malignancies associated with SVCS are either small-cell or non-small-cell lung cancer, an accurate history as to tobacco use is important. A thorough lymph node exam may identify palpable supraclavicular lymphadenopathy or multiple areas of lymphadenopathy, both of which have a high likelihood of being associated with malignancy [19]. Additionally, a thorough physical exam may identify targets that are easily amenable to tissue biopsy in order to establish a diagnosis, as illustrated in the case study.

Radiographic Evaluation

Although the diagnosis is suspected on a clinical basis, it is confirmed by radiologic studies. The chest radiograph (CXR) is abnormal in 84% of patients with SVCS, reflecting the malignancy that is causing the obstruction [2]. Findings on CXR include mediastinal widening (64%) and a superior mediastinal mass, 75% of which occur on the right side, consistent with SVC anatomy (Fig. 22.4). Pleural effusions are found in 25% of patients and, in older radiologic literature, are purported to be found mostly on the right side.



Fig. 22.4 Radiographic evidence of mediastinal mass (lymphoma) with right-sided pleural effusion



Fig. 22.5 Superior vena cava syndrome with blood returning via collaterals draining to the azygos vein. *White open arrows* pointing to collaterals. *White solid arrow* pointing to azygos vein

For definitive diagnosis, computed tomography (CT) of the chest with intravenous contrast is the modality of choice. CT angiography (CTA) can identify the location of the obstruction as above or below the azygos vein. At the same time, it will identify the development of collateral vessels (Fig. 22.5). It can differentiate between intrinsic clot and mass versus extrinsic compression of the superior vena cava with or without a superimposed thrombus [20] (Fig. 22.6). In addition, a chest CT may give information about structures such as the vocal cords/airway, as well as where best to perform a biopsy (via mediastinoscopy, bronchoscopy, or percutaneous fine needle aspirate). CTAs have sensitivities of 96%



Fig. 22.6 Superior vena cava syndrome caused by lung cancer. *White open arrow* pointing to mass obliterating the right main stem bronchus. *White closed arrow* pointing to collapsed superior vena cava

[21]. However, CT imaging and magnetic resonance angiography (MRA) both require the patient to lie flat, which may increase dyspnea and therefore cause movement artifact on images, making interpretation more difficult.

One epidemiologic study found pleural effusions on CT, which occurred in 70% of cases when associated with malignant causes and in only 58% of cases associated with benign causes [22]. Most effusions are small, with the majority filling less than 25% of the hemithorax. Contrary to the expected, pleural effusions occur on either side with equal incidence. The effusions, when sampled, were found to be either chylous in origin or exudative. This differs from the classic thinking that the fluid was transudative due to hydrostatic pressure differences. The chylous origin may be due to obstruction of the thoracic duct or due to impedance of lymphatic flow.

As indwelling devices are a common cause of SVCS, those patients presenting with unilateral arm swelling and an indwelling device will often undergo a Doppler ultrasound of the upper extremity to rule out a thrombus in the subclavian, axillary, and brachiocephalic veins. It is important to note, however, that the SVC cannot be directly imaged by Doppler ultrasound (due to its encasement by the ribs) and therefore additional imaging such as CT is required. Some texts suggest that patency of the SVC can be indirectly determined by normal waveforms in distal veins such as the subclavian or brachiocephalic, but there is no definitive evidence to back this approach. Transesophageal echocardiography would allow for imaging of the SVC and the right atrium but is more invasive.

Contrast venography can also diagnose SVCS, but it requires a significant load of iodinated contrast and radiation

exposure. Venous cannulation of the affected arm may be difficult as well, and if completely obstructed, limited visualization will occur. Finally, extrinsic compression versus internal clot cannot be distinguished. Venography is commonly done before stenting by interventional radiology and will provide information about the patency of the vena cava, as well as the extent of collateral circulation.

Magnetic resonance angiography (MRA) is also relatively sensitive (92–96%) and may be used in those patients with allergy to contrast dye. However, MRA carries the risk of gadolinium-induced nephrogenic systemic fibrosis in patients with renal failure. It provides much greater detailed imaging of the mediastinal structures and can be viewed in multiple planes. Disadvantages include cost and duration of scanning in a dyspneic patient.

Positron emission tomography (PET) scans, either with or without CT scans, are commonly used to assist in the staging of malignant disease both before and after treatment. However, the benefit to diagnosing SVCS with a PET scan is still dependent upon the CT portion of the scan. Specifically, a PET scan without a CT scan will have the ability to show a PET-avid mass in the area of the SVC but will not be able to provide a clear picture of SVC compression or obstruction. Similarly, a PET/CT, with the CT scan performed without IV contrast, will provide better images of the mediastinal area, but diagnosis of SVCS will still be limited. The only way to definitively diagnose SVCS with a PET scan would be to obtain a PET scan followed by a CT scan with IV contrast. Due to the increased costs associated with PET scans and the relative decreased availability, PET scans remain a method of staging prior to initiation of treatment as opposed to a fast or accurate way to diagnose a patient with SVCS.

Histologic Diagnosis

Malignancies remain the most common etiology of SVCS. Malignancy must be confirmed by a tissue diagnosis, although this does not necessarily require a biopsy of the mass that is causing the SVCS. An individualized approach is needed to decide whether less-invasive means can be utilized to obtain a tissue diagnosis before proceeding to mediastinoscopy. Less invasive means of obtaining tissue diagnosis, such as bronchoscopy, transthoracic needle aspiration, palpable lymph node needle aspiration, sputum cytology, or thoracentesis with pleural fluid cytology, are often sufficient for diagnosis. The ideal diagnostic approach is based on underlying patient factors such as ability to tolerate procedure(s), location of disease, and risk of biopsy. For example, if a patient has a palpable supraclavicular or axillary lymph node, needle aspiration biopsy would be preferred as there is high pretest probability and low complication rate. If minimally invasive testing is not diagnostic and a high suspicion for malignancy remains, a mediastinoscopy or mediastinotomy can provide a diagnosis in greater than 90% of the cases, with little increase in complications, as only 3% of patients developed significant hemorrhage with the procedure [23, 24]. Whenever lymphoma is on the differential, an excisional biopsy is preferred in order to allow the pathologist to determine tumor architecture and have sufficient tissue to perform immunohistochemical staining, both of which aid in an accurate diagnosis of the particular subtype of lymphoma. As will be discussed below, most treatments for SVCS are directed toward the causative malignancy, and thus tissue diagnosis is of utmost importance.

Treatment

Historically, treatment for SVCS consisted of emergent radiation, steroids, and diuretics. Importantly, SVCS is described in many textbooks as an oncologic emergency; however, few cases require emergent therapy [25]. In order to attempt to distinguish which patients with SVCS would require aggressive therapy, a proposed staging system has been developed [15]. This algorithm proposes a different treatment approach based on the severity of disease as graded (Table 22.1).

As seen in the treatment algorithm (Fig. 22.7), emergent treatment is indicated only if altered mental status, respiratory distress, or hemodynamic instability is present. The emergent treatment in these cases would be either thrombolytic for thrombus or stent placement for extrinsic compression and/or thrombus. It is important to note that the majority of patients will not present in extremis requiring emergent intervention, allowing for expedited evaluation of the underlying etiology. If not emergent, therapy is supportive or directed at the underlying cause.

Table 22.1 Grading of symptoms associated with SVCS

Grade	Definition	Estimated incidence (%)
0	Asymptomatic; radiographic evidence	10
1	Edema in head/neck, plethora	25
2	Functional impairment of voice, facial muscle by edema (dysphagia, cough, visual disturbance)	50
3	Mild/moderate cerebral edema Mild/moderate laryngeal edema Poor venous return with orthostatic syncope	10
4	Confusion, obtundation Stridor Significant hemodynamic compromise	5
5	Fatal	<1

From Yu et al. [15], with permission Elsevier



Fig. 22.7 Proposed algorithm for treatment of SVCS based on the severity of symptoms and etiology. (From Yu et al. [15], with permission Elsevier)

Supportive Therapy

In the recent past, glucocorticoids have been utilized based on case reports suggesting benefit. However, there is no good evidence to support their use outside of patients for whom lymphoma or thymoma is the cause of SVCS [4]. In a previously undiagnosed patient, this potential benefit should be weighed against the potential risk of eliminating a tissue diagnosis, especially in cases of suspected lymphoma. Diuretics have been used, again based on anecdotal evidence. Since the disease is based on limited flow return to the heart and not on volume overload, it makes little intrinsic sense to utilize diuretics.

Elevation of the head of the bed to theoretically decrease the hydrostatic pressure increase in the brain secondary to the elevated cervical venous pressures has been suggested and has little to no downside.

Treatment of Malignant Causes of SVCS

For patients with SVCS due to malignancy, treatment goals are twofold: (1) relieving the obstruction; and (2) treating the underlying malignancy. Initiation of treatment with chemotherapy and/or radiation therapy before a diagnosis is obtained not only exposes the patient to the side effects of the treatment without guarantee of response but also decreases the yield for an exact tissue diagnosis, with one study reporting only one of six patients (16.7%) having a histologic diagnosis obtained after chemoradiation therapy was initiated [26].

Chemotherapy

Chemotherapy is often the initial treatment of choice for SVCS if the tumor is felt to be chemotherapy sensitive. Small-cell lung cancer, non-Hodgkin lymphomas, and germ cell tumors are all considered chemosensitive, and even symptomatic patients with SVCS will have chemotherapy initiated first. This is in contrast to non-small-cell lung cancer, which traditionally is felt to be less chemosensitive and for whom radiation therapy is often the initial treatment of choice, with or without concurrent chemotherapy [27]. Multiple studies have shown that prompt initiation of chemotherapy in patients with small-cell lung cancer is effective (77% response rate) in alleviating SVCS symptoms within 1-2 weeks of treatment, although a small percentage of patients will experience recurrent obstructive symptoms (17%) [4, 28-32]. Similarly, for lymphomas, prompt initiation of treatment with chemotherapy is usually sufficient to

prevent worsening of symptoms and alleviate obstruction (Fig. 22.8).

With greater understanding of the molecular pathways involved in malignancy, targeted therapies are more commonly employed either alone or in combination with cytotoxic chemotherapy. For example, the BRAF inhibitor, vemurafenib, has been shown to successfully treat SVCS in a patient with melanoma, with resolution of his symptoms by 72 hours [33]. In lung cancer, PD1 or PDL1 inhibitors (pembrolizumab, nivolumab, durvalumab, etc.) are most often given in conjunction with cytotoxic chemotherapy, although monotherapy could be considered in rare situations in which PD1 or PDL1 expression is high and/or the patient has poor functional status.

Radiotherapy

Again, almost 50% of malignancies causing SVCS are nonsmall-cell lung cancer. This tumor is more radiosensitive than chemosensitive; therefore, radiation therapy is the modality of choice for a patient presenting with symptomatic SVCS. In patients receiving radiation, either as monotherapy or in combination with chemotherapy, symptom improvement may begin as soon as 72 hours after initiation of treatment with the majority of responses occurring within 3 weeks. Relapse of SCVS after radiotherapy, chemotherapy, or combined therapy occurred in 16.7% of those with smallcell lung cancer and 20.8% with non-small-cell lung cancer [4]. It should be noted that radiation therapy can paradoxically worsen patients' symptoms with the development of edema associated with treatment. In the long term, radiation-



Fig. 22.8 (a) Superior vena cava syndrome caused by mediastinal tumor (lymphoma). *Black open arrow* pointing to collapsed vena cava with central line in place. (b) Same patient s/p induction chemotherapy

with marked reduction in tumor burden. *Black open arrow* pointing to open vena cava with central line in place

induced fibrosis can result in recurrence of SVC obstruction, although the exact incidence is unknown [34].

However, it is difficult to discern how much of the improvement in symptoms is from the radiation therapy alone versus the development of a collateral vasculature sufficient to compensate for the obstruction. In an interesting study evaluating patients being treated with radiation therapy for SVCS, 85% of the patients reported improvement in symptoms, yet consecutive venograms showed that only 31% of patients had complete resolution of their SVC obstruction. Evaluation during autopsy revealed even lower numbers, with only 24% of patients with either complete or partial resolution of the SVC obstruction [18].

Intravascular Therapy (Stents)

Stents are indicated as an emergent treatment or for cancer that is not responsive to treatment. Intravascular stents can provide symptomatic relief secondary to edema within 48–72 hours. However, one study found that only 17% of patients had complete relief with the stent, suggesting that the edema is multifactorial beyond the vena cava obstruction [35]. Tissue diagnosis will not be adversely affected by stent placement. Stents do not interfere with treatment of the cancer with either radio or chemotherapy and, most importantly,

remain an additional treatment option for those patients who have recurrence of their SVCS after initially responding to either chemotherapy or radiation therapy [35].

Intravascular stents can be placed across the occlusive lesion, thus recanalizing the vein. Catheters are introduced from above via the internal jugular vein or below the diaphragm via the common femoral vein (Fig. 22.9). A femoral approach is beneficial in that it allows for assessment of the brachiocephalic veins and determination of the length of the SVC obstruction [36]. A wire is placed across the obstructive lesion. If recombinant tissue plasminogen activator is to be used, it is often delivered via a catheter with multiple side ports. Then a percutaneous balloon angioplasty is performed with a 10–16 mm angioplasty balloon. After the angioplasty, a stent is placed. The incidence of early re-intervention for thrombus in one study was 4%. Patency at 30 days was 96% [38].

In a study of patients with malignancy-associated SVCS, the treatment was clinically successful in 95% of cases (156 out of 164 patients) with an early mortality rate of 2.4%. Relapse occurred in 22% (36 patients), but re-stenting was successful in 75% of these patients. Recurrence was associated with occlusion, initial associated thrombosis, or the use of steel stents. Complications occurred more commonly if the stent was >16 mm in diameter [37]. Some have suggested using stents for indwelling catheters or primary treatment of benign SVCS [38].



Fig. 22.9 (a) CT coronal view of superior vena cava syndrome. *White open arrow* indicates area of narrowing on contrasted study. (b) Deployment of a stent across the superior vena cava

Complications of stent placement include migration of the stent, hematoma at the site, perforation of the SVC, bleeding, infection, and pulmonary embolism. These have been reported in 3–7% of patients treated with stent placement [2].

Anticoagulation is often recommended after stent placement, but there are no evidence-based protocols to support this practice. Treatment is based upon the use of other stents and the treatment of other venous clotting. Typical practice is to use anticoagulation for 3–4 days post procedure with consideration for long-term anticoagulation based on an individualized assessment of risk factors for rethrombosis [36].

Treatment of Benign (Nonmalignant Causes) of SVCS

Medical treatment with steroids or diuretics has not been shown to be useful, especially in treating SVCS of benign origin. If an indwelling line is present, and the SVC was due to a clot, then anticoagulation is often utilized, although the effectiveness of this approach has not been shown either in the short- or long-term setting.

Thrombolytics

Patients who have developed SVCS due to intraluminal thrombus may benefit from thrombolysis, often in combination with stent placement. The use of thrombolytics (typically tissue plasminogen activator or urokinase) before stents are placed has been shown to decrease the length of blockage due to thrombus and therefore the number of stents placed [35, 39–43]. It is also felt that thrombolysis prior to stent placement decreases the burden of material that has the potential to embolize during stent placement. Adverse events, however, are increased as well when thrombolytics are used, including complications such as gastrointestinal hemorrhage, hemoptysis, and intracranial hemorrhage. This does have to be balanced with the relative success rate of thrombolysis, which has been reported to be as high as 88% [44]. Although typically thrombolysis occurs by pharmacological means, new advances in technology have provided additional methods of clot dissolution, primarily through the use of ultrasound-accelerated catheter-directed thrombolysis, which utilizes high-frequency sound waves in combination with tissue plasminogen activator [45]. This relatively new technology will require additional research to determine its efficacy relative to traditional chemical thrombolysis.

Prognosis

The outcomes of patients with malignant SVCS are directly related to the underlying malignancy and do not necessarily portend a lower overall survival compared to patients with similar tumor types who do not have SVCS [2]. Within the literature for small-cell lung cancer alone, studies have shown either no change or an improvement in overall survival for those patients who presented with SVCS when compared to patients who did not develop SVCS [28, 30–32, 46]. This may be due to patients with SVCS symptoms presenting to medical care earlier in the course of disease.

While many studies report the median life expectancy for a patient with SVCS as only 6 months, patients may survive over 2 years after treatment [15]. In some patients, treatment of the underlying malignant disease will result in cure of the disease and resolution of the SVCS.

In addition to the underlying etiology, poor prognostic factors for SVCS in patients with malignant cause include advanced age (>50), history of smoking, and use of steroids [47].

Recurrence (Durability of Treatment)

Almost 32% of patients with SVCS secondary to small-cell lung cancer, after treatment with chemotherapy, radiotherapy, or indeed both, have a recurrence of SVCS; however, these data are from 1983 and treatment advances may have improved this prognosis [29]. Relapse after placement of an SVC stent is approximately 11% (reported values of 9–20%). Most of these undergo a successful second stent placement, although a small percentage of patients will have recurrent obstructive symptoms (17%) [4, 28–32].

Palliative Care Discussions

As with any potentially life-threatening illness, palliative care considerations are a must. There are many definitions of palliative care, but most involve the concepts of preventing and minimizing suffering, optimizing one's quality of life, and aligning healthcare with patient goals and values. Many organizations, including the Institute of Medicine and the World Health Organization, recognize that palliative care specialists are integral to superior cancer care [48]. One widely quoted study identified an improved quality of life for patients with metastatic non-small-cell lung cancer who received early palliative care interventions while also receiving standard oncologic care [49]. The American Society of Clinical Oncologic clinic practice guidelines also recom-

mends involving palliative care specialists early in the disease course of patients with advanced cancer, in combination with chemotherapy and/or radiation therapy [50].

Many have used the concept of estimated life expectancy as a surrogate marker for when to more fully involve the palliative care team in a patient's care. In fact, the National Comprehensive Cancer Network's Guidelines for Palliative Care references a life expectancy of less than 6 months as a trigger to more actively engage palliative care, for which SVCS is one indicator [51]. However, additional factors must be taken into account, including the underlying malignancy, the patient's performance status, and patient goals of care. It is crucial that an oncologic specialist be involved in the conversations with the patient and other members of the healthcare team in order to provide an overall perspective of the disease course and potential outcomes.

References

- 1. Hunter W. The history of an aneurysm of the aorta with some remarks on aneurysm in general. Med Observ Inq. 1757;1:34.
- Wilson LD, Detterbeck FC, Yahalom J. Clinical practice. Superior vena cava syndrome with malignant causes. N Engl J Med. 2007;356(18):1862–9. https://doi.org/10.1056/NEJMcp067190.
- Rice TW, Rodriguez RM, Light RW. The superior vena cava syndrome: clinical characteristics and evolving etiology. Medicine (Baltimore). 2006;85(1):37–42.
- Rowell NP, Gleeson FV. Steroids, radiotherapy, chemotherapy and stents for superior vena caval obstruction in carcinoma of the bronchus: a systematic review. Clin Oncol (R Coll Radiol). 2002;14(5):338–51.
- Lazzarino M, Orlandi E, Paulli M, Boveri E, Morra E, Brusamolino E, et al. Primary mediastinal B-cell lymphoma with sclerosis: an aggressive tumor with distinctive clinical and pathologic features. J Clin Oncol. 1993;11(12):2306–13.
- Perez-Soler R, McLaughlin P, Velasquez WS, Hagemeister FB, Zornoza J, Manning JT, et al. Clinical features and results of management of superior vena cava syndrome secondary to lymphoma. J Clin Oncol. 1984;2(4):260–6.
- Savarese DM, Zavarin M, Smyczynski MS, Rohrer MJ, Hutzler MJ. Superior vena cava syndrome secondary to an angiotropic large cell lymphoma. Cancer. 2000;89(12):2515–20.
- Kurata A, Saji H, Ikeda N, Kuroda M. Intracaval and intracardiac extension of invasive thymoma complicated by superior and inferior vena cava syndrome. Pathol Int. 2013;63(1):56–62.
- Behl D, Hendrickson AW, Moynihan TJ. Oncologic emergencies. Crit Care Clin. 2010;26(1):181–205.
- Cheng S. Superior vena cava syndrome: a contemporary review of a historic disease. Cardiol Rev. 2009;17(1):16–23.
- Rossi A, Baravelli M, Cattaneo P, Romano M, Maricalco G, Imperiale D, et al. Acute superior vena cava syndrome after insertion of implantable cardioverter defibrillator. J Interv Card Electrophysiol. 2008;23(3):247–9.
- Barakat K, Robinson NM, Spurrell RA. Transvenous pacing leadinduced thrombosis: a series of cases with a review of the literature. Cardiology. 2000;93(3):142–8.
- Gonzalez-Fajardo JA, Garcia-Yuste M, Florez S, Ramos G, Alvarez T, Coca JM. Hemodynamic and cerebral repercussions arising from surgical interruption of the superior vena cava. Experimental model. J Thorac Cardiovasc Surg. 1994;107(4):1044–9.

- Drews RE, Rabkin DJ. Malignancy-related superior vena cava syndrome. In: Post TW, editor. UpToDate. Waltham: UpToDate; 2020.
- Yu JB, Wilson LD, Detterbeck FC. Superior vena cava syndrome a proposed classification system and algorithm for management. J Thorac Oncol. 2008;3(8):811–4. https://doi.org/10.1097/ JTO.0b013e3181804791.
- McCurdy MT, Shanholtz CB. Oncologic emergencies. Crit Care Med. 2012;40(7):2212–22. https://doi.org/10.1097/ CCM.0b013e31824e1865.
- Crispo MM, Fidalgo G, Fix ML, Higgins GL 3rd. A case of superior vena cava syndrome demonstrating Pemberton sign. J Emerg Med. 2012;43(6):1079–80.
- Ahmann FR. A reassessment of the clinical implications of the superior vena caval syndrome. J Clin Oncol. 1984;2(8):961–9.
- Chau I, Kelleher MT, Cunningham D, Norman AR, Wotherspoon A, Trott P, et al. Rapid access multidisciplinary lymph node diagnostic clinic: analysis of 550 patients. Br J Cancer. 2003;88(3): 354–61.
- Katabathina VS, Restrepo CS, Betancourt Cuellar SL, Riascos RF, Menias CO. Imaging of oncologic emergencies: what every radiologist should know. Radiographics. 2013;33(6):1533–53.
- Warren P, Burke C. Endovascular management of chronic upper extremity deep vein thrombosis and superior vena cava syndrome. Semin Intervent Radiol. 2011;28(1):32–8.
- Rice TW. Pleural effusions in superior vena cava syndrome: prevalence, characteristics, and proposed pathophysiology. Curr Opin Pulm Med. 2007;13(4):324–7.
- 23. Dosios T, Theakos N, Chatziantoniou C. Cervical mediastinoscopy and anterior mediastinotomy in superior vena cava obstruction. Chest. 2005;128(3):1551–6.
- Mineo TC, Amrogi V, Nofroni I, Pistolese C. Mediastinoscopy in superior vena cava obstruction: analysis of 80 consecutive patients. Ann Thorac Surg. 1999;68(1):223–6.
- Schraufnagel DE, Hill R, Leech JA, Pare JA. Superior vena caval obstruction. Is it a medical emergency? Am J Med. 1981;70(6):1169–74.
- Bagheri R, Rahim M, Rexaeetalab F, Akbari H, Shojaian R. Malignant superior vena cava syndrome: is this a medical emergency? Ann Thorac Cardiovasc Surg. 2009;15(2):89–92.
- 27. Simoff MJ, Lally B, Slade MG, Goldberg WG, Lee P, Michaud GC, et al. Symptom management in patients with lung cancer diagnosis and management of lung cancer. 3rd ed. American College of Chest Physicians evidence-based clinical practice guidelines. Chest. 2013;143(5):E455–97.
- Dombernowsky P, Hansen HH. Combination chemotherapy in the management of superior vena caval obstruction in small-cell anaplastic carcinoma of the lung. Acta Med Scand. 1978;204(6):513–6.
- 29. Spiro SG, Shah S, Harper PG, Tobias JS, Geddes DM, Souhami RL. Treatment of obstruction of the superior vena cava by combination chemotherapy with and without irradiation in small-cell carcinoma of the bronchus. Thorax. 1983;38(7):501–5.
- Sculier JP, Evans WK, Feld R, DeBoer G, Payne DG, Shepherd FA, et al. Superior vena caval obstruction syndrome in small cell lung cancer. Cancer. 1986;57(4):847–51.
- Urban T, LeBeau B, Chastang C, Leclerc P, Botto MJ, Sauvaget J. Superior vena cava syndrome in small-cell lung cancer. Arch Intern Med. 1993;153(3):384–7.
- 32. Wurschmidt F, Bunemann H, Heilmann HP. Small cell lung cancer with and without superior vena cava syndrome: a multivariate analysis of prognostic factors in 408 cases. Int J Radiat Oncol Biol Phys. 1995;33(1):77–82.
- Mesko SM, Rosenthal KJ, Boasberg PD, Hamid O. BRAF-targeted therapy to treat superior vena cava syndrome in a patient with metastatic cancer. J Clin Oncol. 2015;33(25):e101–3.

- Van Putten JW, Schlosser NJ, Vujaskovic Z, Leest AH, Groen HJ. Superior vena cava obstruction caused by radiation induced venous fibrosis. Thorax. 2000;55(3):245–6.
- 35. Tanigawa N, Sawada S, Mishima K, Okuda Y, Mizukawa K, Ohmura N, et al. Clinical outcome of stenting in superior vena cava syndrome associated with malignant tumors. Comparison with conventional treatment. Acta Radiol. 1998;39(6):669–74.
- Friedman T, Quencer KB, Kishore SA, Winokur RS, Madoff DC. Malignant venous obstruction: superior vena cava syndrome and beyond. Semin Intervent Radiol. 2017;34(4):398–408.
- 37. Fagedet D, Thony F, Timsit JF, Rodiere M, Monnin-Bares V, Ferretti GR, et al. Endovascular treatment of malignant superior vena cava syndrome: results and predictive factors of clinical efficacy. Cardiovasc Intervent Radiol. 2013;36(1):140–9.
- Rizvi AZ, Kalra M, Bjarnason H, Bower TC, Schleck C, Gloviczki P. Benign superior vena cava syndrome: stenting is now the first line of treatment. J Vasc Surg. 2008;47(2):372–80.
- Crowe MT, Davies CH, Gaines PA. Percutaneous management of superior vena cava occlusions. Cardiovasc Intervent Radiol. 1995;18(6):367–72.
- Edwards RD, Jackson JE. Case report: superior vena caval obstruction treated by thrombolysis, mechanical thrombectomy and metallic stents. Clin Radiol. 1993;48(3):215–7.
- Kee ST, Kinoshita L, Razavi MK, Nyman UR, Semba CP, Dake MD. Superior vena cava syndrome: treatment with catheter-directed thrombolysis and endovascular stent placement. Radiology. 1998;206(1):187–93.
- Lepper PM, Ott SR, Hopper H, Schumann C, Stammberger U, Bugalo A, et al. Superior vena cava syndrome in thoracic malignancies. Respir Care. 2011;56(5):653–66.

- Uberoi R. Quality assurance guidelines for superior vena cava stenting in malignant disease. Cardiovasc Intervent Radiol. 2006;29(3):319–22.
- Ostler PJ, Clarke DP, Watkinson AF, Gaze MN. Superior vena cava obstruction: a modern management strategy. Clin Oncol (R Coll Radiol). 1997;9(2):83–9.
- 45. Dumantepe M, Tarhan A, Ozler A. Successful treatment of central venous catheter induced superior vena cava syndrome with ultrasound accelerated catheter-directed thrombolysis. Catheter Cardiovasc Interv. 2013;81(7):E269–73.
- 46. Warde P, Payne D. Does thoracic irradiation improve survival and local control in limited-stage small-cell carcinoma of the lung? A meta-analysis. J Clin Oncol. 1992;10(6):890–5.
- 47. Chan RC, Chan YC, Cheng SW. Mid- and long-term follow-up experience in patients with malignant superior vena cava obstruction. Interact Cardiovasc Thorac Surg. 2013;16(4):455–8.
- Ferris FD, Bruera E, Cherny N, Cummings C, Currow D, Dudgeon D, et al. Palliative cancer care a decade later: accomplishments, the need, next steps—from the American Society of Clinical Oncology. J Clin Oncol. 2009;27(18):3052–8.
- 49. Temel JS, Greer JA, Muzikansky A, Gallagher ER, Admane S, Jackson VA, et al. Early palliative care for patients with metastatic non-small-cell lung cancer. N Engl J Med. 2010;363(8):733–42.
- Ferrell BR, Temel JS, Temin S, Alesi ER, Balboni TA, Basch EM, Firn JI, et al. Integration of palliative care into standard oncology care: American Society of Clinical Oncology clinical practice guideline update. J Clin Oncol. 2017;35(1):96–112.
- Levy MH, Adolph MD, Back A, Block S, Codada SN, Dalal S, et al. Palliative care. J Natl Compr Cancer Netw. 2012;10(10):1284–309.



Nagendra Y. Madisi and Wissam Abouzgheib



23

Case Study

A 73-year-old male with a remote history of smoking presented with complaints of worsening shortness of breath and dry cough. His imaging studies showed a large right pleural effusion, and a PET scan revealed extensive pleural hypermetabolism throughout the right hemithorax, suspicious for malignancy. Bronchoscopy and transbronchial needle aspiration were nondiagnostic. Pathological results of a pleural biopsy and pleural fluid were positive for Ber-EP4. MOC31. p53, and CK7 consistent with adenocarcinoma with lung primary. The patient was treated with one cycle of combination chemotherapy following which immunohistochemical staining indicated that 70% of the patient's tumor cells expressed PD-L1. The patient began a 3-week cycle of pembrolizumab. Initially, pembrolizumab was tolerated fairly well, but after completing four cycles of single-agent therapy the patient developed increasing dyspnea and CT chest showed patchy areas of ground-glass opacity throughout both lungs. Based on these imaging findings, the clinical course, as well as negative results of cultures and viral antibody tests, the patient was diagnosed with pembrolizumab-induced pneumonitis. Thereafter, immune checkpoint inhibitor was held and prednisone 40 mg per day with a slow taper begun. This regimen resulted in the regression of both the pulmonary opacities and symptoms.

Introduction

The worldwide surveillance of cancer survival during a 15-year period (2000–2014) was recently reported from 71 countries, including data on 18 cancer types. While cancer incidence is increasing, survival trends are reported to be increasing, even for some of the more lethal cancers [1]. This

N. Y. Madisi · W. Abouzgheib (🖂)

is directly linked to an aging population, improved diagnostics and screening tools for cancer, high population awareness, and more advanced therapeutic options. The incidence rate of new cancer cases in the United States is 436 per 100,000 population [2]. In 2019, there were an estimated 1.7 million new cancer cases diagnosed and 606,880 cancer deaths in the United States [2]. Intensive chemotherapy regimens and the use of new and more targeted therapeutic drugs have resulted in higher cancer cure rates. However, the treatment often leads to repeated invasive procedures, drugrelated organ toxicities, and increased susceptibility to infection. As a consequence, emergency physicians and intensivists are increasingly managing cancer patients presenting with single- or multi-organ dysfunction. Pulmonary complications include respiratory failure, including acute respiratory distress syndrome (ARDS); pleural diseases (such as pleural effusion or pneumothorax); chemotherapy or radiation-induced pulmonary toxicity; hemoptysis; and pulmonary embolism.

Respiratory Failure, ARDS, and Ventilator Management

Respiratory Failure

Acute respiratory failure (ARF) is common in cancer patients, occurring in up to 50% of hematological malignancies and 15% of solid tumors or solid organ transplantation. It is associated with high mortality rate. Risk factors include invasive mechanical ventilation, organ dysfunction, advanced age, poor performance status, delayed intensive care unit admission, and invasive fungal infection [3]. Etiologies of respiratory failure in cancer patients are numerous. The most frequent etiologies include pneumonia, cardiogenic pulmonary edema, ARDS, chemotherapy or radiation-induced lung injury, pneumothorax and bronchopleural fistula, large pleural effusions, hemoptysis, and thrombotic/nonthrombotic embolus [4–6].

Division of Pulmonary and Critical Care, Department of Medicine, Cooper Medical School at Rowan University, Camden, NJ, USA e-mail: abouzgheib-wissam@cooperhealth.edu

Acute Respiratory Distress Syndrome (ARDS)

ARDS represents an acute and diffuse inflammatory lung injury. It leads to increased pulmonary vascular permeability, acute inflammation of the alveolar walls, and diffuse alveolar damage [3, 7]. Clinical hallmarks of ARDS are hypoxemia and bilateral radiographic opacities in the absence of heart failure. The hypoxemia is profound as defined by a ratio of arterial oxygen to fraction of inspired oxygen (PaO₂/FiO₂) of 300 or less. ARDS is further subdivided into mild (300–201 ratio), moderate (200–100 ratio), and severe (≤ 100 ratio) [8].

ARDS in patients with malignancies is not well studied, as these patients are often excluded from ARDS trials; however, a few studies are available [9, 10]. ARDS incidence is reported around 12% in all cancer patients in the ICU and 18% among those on mechanical ventilation. ARDS is more common in hematologic malignancies. Higher mortality was noted overall in retrospective assessment of ARDS network trials and was attributed to a more severe presentation and advanced age [11-13]. The mortality of cancer patients with ARDS in these trials was 54% compared to 24% among noncancer patients [13]. In a more recent prospective study [10], the main factors associated with higher severity and mortality were the duration of mechanical ventilation and ICU stay prior to ARDS onset. Later onset was associated with higher mortality. Other risk factors included excessive positive fluid balance before ARDS onset and acute kidney injury. The presence of neutropenia is also an important contributing factor, whether a manifestation of the underlying cancer or secondary to chemotherapy. The course of ARDS in patients with neutropenia was different from that of the general population. In general, the outcome of ARDS is determined in the first 10 days, by which time half of the patients either have died or weaned off treatments. In patients with neutropenia, more than 85% of ICU survivors were still hospitalized after 10 days [14]. One recent study evaluated the various etiologies of ARDS in patients with malignancies; infectious etiologies were found in almost 90% of patients [15, 16].

In the same study [15], noninvasive positive-pressure ventilation (NIPPV) was used in more than a third of cancer patients with ARDS. Ultimately, the majority (71%) of patients on NIPPV required endotracheal intubation in correlation with severity of ARDS. Failure of NIPPV ventilation was associated with worse outcome. This study also examined potential prognostic indicators and found that two factors are associated with lower hospital mortality: solid tumor and primary ARDS (caused by direct lung insult, including infectious or noninfectious causes). Factors associated with higher mortality are allogenic bone marrow transplant, higher admission Sequential Organ Failure Assessment scores (SOFA) [17], presence of invasive fungal infection, and failure of NIPPV [4, 13].

Ventilator Management

Early recognition in the emergency department (ED) and initiation of supportive therapy, including mechanical ventilation, are mainstays in the management of ARDS patients.

ED length of stay for patients requiring admission to intensive care units has increased gradually in recent years [18]. Mechanical ventilation is an integral part of critical care and mechanically ventilated patients are commonly managed and monitored by emergency physicians. With the increasing demand for ED care as well as ICU beds [19], emergency physicians are expected to manage many patients on mechanical ventilation for prolonged periods of time. When invasive mechanical ventilation is initiated and managed in the ED, the emergency physician should have an understanding of open-lung ventilation and associated low tidal volume ventilation (LTVV) or lung-protective positivepressure ventilation.

NIPPV delivered to selected patients obviates the need for an endotracheal tube. The ventilator is connected to the patient via a facemask. The mask is attached firmly to the patient's face using straps in order to prevent air leak. Most clinicians prefer the use of invasive intermittent positivepressure ventilation (IPPV) over NIPPV in patients with ARDS considering the potential for hypoxemia to worsen in these patients and the risk for rapid deterioration. Other noninvasive ventilation strategies include high-flow nasal cannula (HFNC) oxygen therapy that delivers a high flow of heated and humidified gas. In a recent post hoc analysis including 82 immunocompromised patients, those treated with HFNC alone had a lower intubation rate than the NIPPV group (31% with HFNC vs. 65% with NIPPV). Similarly, the mortality rate at day 90 was lower with HFNC compared to NIPPV [20]. These differences were not observed between HFNC and the standard oxygen group. A prospective randomized study evaluated the early use of NIPPV versus high concentration of oxygen in a less severe group of patients with mild ARDS [21]. Despite a historical high rate of intubation in patients with ARDS initiated on NIPPV, this study showed a significant decrease in the respiratory rate, improved PaO₂/FiO₂ ratio, and lower incidence of subsequent organ failure. However, this study had a very selective young group of patients who were able to tolerate and cooperate with this mode of ventilation.

When using IPPV, it is very important to consider the following issues: (a) alveolar involvement in ARDS is heterogeneous; and (b) damage caused by adjustments in ventilation to maintain adequate blood gases may result in delayed additive iatrogenic lung injury. Therefore, LTVV (low tidal volume ventilation) is the preferred mode of ventilation for patients with ARDS. The rationale for this mode is that overdistension of the alveoli is a major reason for ventilator-induced lung



Fig. 23.1 The inspiration limb of a pressure–volume curve in an ARDS patient without PEEP application. Initial pressure application produces very little lung inflation as the pressure is applied against wet boggy lung. As lung begins to inflate, compliance improves and the curve assumes a steeper slope. As tidal volumes become too large for a lung that is questionable for ventilation (overinflation), the curve flattens. Low tidal volume strategy in ARDS should target avoiding moving into the flattening portion of curve (upper deflection zone, *upper arrow*). The goal of PEEP application is to avoid lung collapse at end inspiration and a repeated cycle of lung collapse and reopening with each delivered ventilation breath (with production of shear force injury or atelectrauma). Therefore, optimal PEEP would be applied at the lower end of the upslope of the curve (lower arrow, lower inflation point)

injury (Fig. 23.1). The majority of evidence suggests that LTVV improves mortality as well as other meaningful outcomes in patients with ARDS. The multicenter ARMA trial [22] compared LTVV (initial tidal volume 6 ml/kg predicted body weight, PBW) versus conventional ventilation (initial tidal volume 12 ml/kg/PBW). The benefits of LTVV were lower mortality rates (31 vs. 40%) and more ventilator-free days (12 vs. 10 days). As expected, LTVV may be associated with hypercapnia, which is generally well tolerated and may be associated with beneficial effects not directly related to LTVV [23, 24]. In permissive hypercapnia, the accepted and managed rise of PaCO₂ and subsequent acidosis increases arterial and tissue oxygenation by a right shift of the oxygenhemoglobin dissociation curve and possibly by increasing cardiac output and circulating catecholamines. Hypercapnic acidosis reduces cyclic mechanical stretch-induced nuclear factor-kB activation, reduces interleukin-8 production, and decreases epithelial injury and cell death compared to normocapnia [24, 25]. However, the rise of $PaCO_2$ should occur gradually. Rapid rise should be avoided as the negative effects may exceed the beneficial ones (increased heart rate/blood pressure, arrhythmias, and pulmonary vasoconstriction/worsening hypoxemia). A typical approach for enacting a low tidal volume strategy in ARDS is as follows: (a) set tidal volume initially to 8 ml/kg/PBW; (b) titrate down to 7 and then 6 ml/ kg/ PBW; (c) measure the airway plateau pressure (Pplat), if

Table 23.1 ARDSnet PEEP table

Lower PEEP/higher FiO ₂												
FiO ₂	0.3	0.4	0.4	0.5	0.5	0.6	0.7	0.7				
PEEP	5	5	8	8	10	10	10	12				
FiO ₂	0.7	0.8	0.9	0.9	0.9	1.0						
PEEP	14	14	14	16	18	18-24						
Higher PEEP/lower FiO ₂												
FiO ₂	0.3	0.3	0.3	0.3	0.3	0.4	0.4	0.5				
PEEP	5	8	10	12	14	14	16	16				
FFiO ₂	0.5	0.5-0.8	0.8	0.9	1.0	1.0						
PEEP	18	20	22	22	22	24						

From NIH NHLBI ARDS Clinical Network Mechanical Ventilation Protocol Summary. http://www.ardsnet.org/files/ventilator_protocol_ 2008-07.pdf

Use a minimum PEEP of 5 cmH_2O . Consider use of incremental FiO₂/ PEEP combinations such as shown below to achieve goal. Consider the higher PEEP table in the presence of more severe hypoxemia

PEEP Positive end-expiratory pressure, FiO_2 Fraction of inspired oxygen

 \leq 30 cmH₂O, no other adjustment is required; and (d) if Pplat is >30 cmH₂O, then further decrease the tidal volume to as low as 4 ml/kg/PBW to achieve target. Higher Pplat may be allowed in the presence of obesity or anasarca ng volume (mL)

Open-lung ventilation represents the addition of positive end-expiratory pressure (PEEP) to the LTVV strategy, targeting to prevent collapse of edematous lung at end expiration. PEEP is believed to maximize alveolar recruitment and prevent cyclic atelectasis. According to several metaanalyses, the use of open-lung ventilation is associated with improved oxygenation; however, the effect on mortality has not been well established. The ARMA trial used a type of open-lung strategy in both arms, increasing PEEP levels with increasing severity of hypoxemia. Open-lung strategies have repeatedly shown improved oxygenation and perhaps more importantly, improved lung compliance. The titration of PEEP is typically based on oxygenation deficit or pressurevolume curves (see discussion below). When guided by oxygenation, start with the lowest PEEP possible to maintain an adequate PaO₂ of 55-80 mmHg with a FiO₂ of less than 60% and then titrate PEEP according to the ARDS net PEEP/FiO₂ protocol (Table 23.1) (http://www.ardsnet.org/files/ventilator protocol 2008-07.pdf). When pressure-volume curves are used, it is important to calculate lung compliance and use a PEEP level that moves the end-expiratory P/V point onto the steep part of the pressure-volume curve (see Fig. 23.1).

One additional, nonventilation-related strategy is worthy of mention in the early management of ARDS patients, that is, the use of neuromuscular blockade. There is some evidence that early use of neuromuscular blockade agents in patients with ARDS is associated with better outcomes, including mortality [26, 27]. The recent Reevaluation of Systemic Early Neuromuscular Blockade (ROSE) trial involved patients with ARDS who were treated with a strategy involving a high PEEP and who showed no significant difference in mortality at 90 days between the cisatracurium group and the control group [28]. This finding is contrary to the results of the *ARDS et Curarisation Systématique* (ACURASYS) trial published in 2010, in which the adjusted 90-day mortality was lower in the cisatracurium group than in the placebo group [20, 26, 27, 29].

In the paralyzed patient, both the Pplat, as an estimate of end-inspiratory pressure, and the pleural pressure Ppl are positive; therefore, the TP pressure estimate is Pplat – Ppl. In the spontaneous breathing patient, the pleural pressure may be negative at end inspiration, and this negative pressure will increase transpulmonary pressure with the delivery of the same tidal volume. Paralysis, by eliminating inspiratory effort, would be expected to decrease TP pressure in the presence of overinflation, thereby decreasing the risk of ventilator-induced lung injury.

Summary

Acute respiratory failure and ARDS are common in patients with malignancies. Infectious etiologies are most common. Early recognition and intervention are crucial and should be initiated in the ED upon presentation. A trial of NIPPV/ HFNC is acceptable initially in stable and cooperative patients. Lung-protective and open-lung ventilation strategies are keys to improve outcomes and survival. Given the limited treatment options for ARDS, and the early onset after admission, measures to prevent onset and mitigate severity should be instituted in the ED.

Pneumothorax and Pleural Effusion

Pleural manifestations are not uncommon in patients with malignancies. The pleura is often a metastasis site from local or distant cancers, presenting more commonly as pleural effusions rather than solid masses. Also, the pleura can be involved with spontaneous or iatrogenic pneumothoraces in patients with malignancies.

Pneumothorax

Definitions, Etiologies, and Diagnostic Modalities

Pneumothorax refers to the presence of air in the pleural space. The classification of pneumothorax includes spontaneous, traumatic, or iatrogenic. Spontaneous pneumothorax occurs without obvious cause, either primary without evidence of underlying lung disease, or secondary with apparent underlying lung disease, often COPD. Traumatic pneumothorax occurs after blunt or penetrating trauma to the chest. Iatrogenic pneumothorax occurs after a diagnostic or therapeutic intervention, such as transthoracic lung biopsy, central line placement, or barotrauma, due to mechanical ventilation. The incidence of pneumothorax in patients receiving mechanical ventilation ranges between 7% and 14% [30]. Patients with acute lung injury or ARDS are at increased risk.

Clinical manifestations are widely variable, ranging from asymptomatic to respiratory failure to prolonged bronchopleural fistulas (BPFs). Prompt diagnosis and management are crucial especially in symptomatic patients with underlying lung diseases or critically ill patients requiring mechanical ventilation.

Sometimes found incidentally on routine chest imaging, the presence of a pneumothorax is clinically suspected in the appropriate clinical setting. A small pneumothorax can be asymptomatic and self-limited, whereas a large pneumothorax can cause hypoventilation, hypoxemia, and/or hemodynamic instability.

Tension pneumothorax represents a surgical emergency and requires emergent intervention. It may lead to respiratory failure requiring mechanical ventilation. It may also complicate preexisting respiratory failure on positivepressure ventilation.

In intubated and sedated patients, a pneumothorax should be suspected with sudden and unexplained worsening respiratory failure, increased oxygen requirements, hemodynamic instability, and a sudden rise in peak and plateau pressures. It is frequently diagnosed based on clinical presentation, risk factors, and physical exam (not by imaging) and should be treated by immediate emergent decompression in hemodynamically unstable patients. However, the increasing availability of bedside ultrasonography has made emergent imaging confirmation of pneumothoraces possible prior to emergent decompression.

The current first-line imaging modality used to identify a pneumothorax is chest radiography, although bedside ultrasonography has great potential to compete as the diagnostic method of choice in the future. The typical finding is displacement of the white visceral pleural line from the chest wall on an upright chest radiograph. The underlying lung parenchyma should be examined for the presence of lung disease that might suggest a cause for the pneumothorax. In bedridden or ICU patients, care should be exercised in order to differentiate the visceral pleural line from skin folds. Skin folds frequently extend beyond the rib cage; blood vessels and lung parenchyma often extend beyond the skin fold. Their attenuation profile is also different, forming a negative black Mach band instead of the white visceral pleural line (Fig. 23.2).

Computed tomography (CT) diagnosis is best utilized for complicated or unclear situations. CT scans are more accurate in determining size of pneumothorax when compared to chest radiography [31]. Fig. 23.2 (a) White visceral pleural line in pneumothorax. (b) Black Mach band in skin fold

307



Point-of-care ultrasound is increasingly used at the bedside and is the modality of choice in the ICU and ED to integrate the clinical assessment of the critically ill: in particular, lung ultrasound has advanced significantly in the last decade. Lung ultrasound can be used for early detection and management of respiratory complications under mechanical ventilation, such as pneumothorax, ventilatorassociated pneumonia, atelectasis, and pleural effusions [31, 32]. Beside ultrasonography offers several advantages over chest radiography or CT scans including rapid availability, lack of radiation, real-time interpretation, and lower cost. It also offers the ability to immediately rule out a pneumothorax after an invasive procedure or in the midst of a clinical deterioration.

Clinical Scenarios in Cancer Patients

The common ED clinical scenarios involving a cancer patient presenting with a pneumothorax include the following:

1. Secondary spontaneous pneumothorax in patients with underlying lung disease (e.g., emphysema) and a concomitant cancer diagnosis: In fact, most cancer patients presenting with a secondary spontaneous pneumothorax had lung cancer (20%) or underlying chronic obstructive pulmonary disease (COPD or emphysema) (52%) [33]. The mechanisms could be due to bronchopleural fistula within a necrotic tumor, tumor-induced rupture of a subpleural bleb, or direct invasion of the pleura. These patients require hospitalization for observation, and potentially, chest tube insertion. Definitive surgical interventions, such as pleurodesis or wedge resection, may be required. As would be expected, patients with active cancer have a significantly worse survival compared to those without active cancer (3 vs. 113 months) [33].

- 2. Iatrogenic pneumothorax following diagnostic procedures: CT-guided or bronchoscopic lung biopsies, as well as therapeutic procedures, such as thoracentesis, bronchoscopic endobronchial tumor ablations, or percutaneous radiofrequency ablation, may produce pneumothorax. CT-guided biopsy is often considered when the abnormality is not easily accessible with a bronchoscope but the procedure is associated with higher rate of complications. For CT-guided core biopsy, the pooled rate of pneumothorax was 25.3%, pulmonary hemorrhage 18.0%, and hemoptysis 4.1%. For FNA procedures, these rates were lower, 18.8%, 6.4%, and 1.7%, respectively [34]. A high level of suspicion should exist in patients who undergo CT-guided biopsy and present with worsening dyspnea, cough, chest pressure, or pain. Usually symptoms occur within 3 h after the procedure; however, onset can be delayed in a small percentage (<4%) of patients [35]. Approximately 20% of patients with this complication require chest tube insertion, depending on the size of the pneumothorax and associated symptoms. Predisposing factors include distance of the target to the pleura, number of needle passes traversing normal lung, and presence of underlying lung diseases [36].
- 3. Treatment-related pneumothorax secondary to thoracentesis, bronchoscopic ablation of central airway obstrucpercutaneous radiofrequency ablation, tion, orchemo-induced tumor necrosis: With the use of ultrasound, the rate of pneumothorax post-thoracentesis has dropped to less than 0.63% [35]. When a pneumothorax occurs, it is usually small and only a third of patients

require chest tube insertion. A chest tube should be considered if the pneumothorax is large or progressive and the patient is symptomatic or requires mechanical ventilation.

Central airway malignant obstruction is currently aggressively treated with several minimally invasive bronchoscopic interventions. One third of patients with advanced lung cancer develop central airway obstruction, and several other malignancies may metastasize to the airway. Pneumothorax after an ablative bronchoscopy, although rare, is usually immediate and treated in the bronchoscopy suite. It is most often related to supportive care, such as jet ventilation during rigid bronchoscopy. Nevertheless, pneumothorax must be ruled out in symptomatic patients presenting after any ablative bronchoscopy.

Image-guided percutaneous therapies became popular over the last two decades for the treatment of pulmonary malignancies, especially in nonsurgical candidate patients. They include radiofrequency or microwave ablation. Pneumothorax, pleural effusion, and subcutaneous emphysema were observed after radiofrequency ablation, respectively, in 15.9%, 20%, and 5% of patients in a recent study by Picchi et al. [37].

Pneumothorax is the most common peri-procedural complication after microwave ablation with an incidence of 33.9% and only 11% of these patients require intervention [37, 38]. Risk factors are similar to CT-guided biopsies. In addition, the number of tumor ablations during a single procedure and lack of prior lung surgery are risk factors as well. Patients with underlying emphysema are at increased risk of pneumothorax [39].

Treatment

Simple manual aspiration using an intercostal needle or small catheter is indicated in noncomplicated patients presenting with the first episode of a large or symptomatic secondary spontaneous pneumothorax. The most recent Cochrane review concluded higher immediate success rates in the needle aspiration (NA) group compared to chest tube drainage and shorter hospital length of stay in the NA group [40]. Treatment consists of inserting a needle or catheter in the pleural space, aspirating the pleural air, followed by removal of the needle or catheter. The resolution rate is high [41]. When simple aspiration is unsuccessful to keep the lung inflated or when air leak is large or persistent, a tube thoracostomy is indicated. There is no evidence that large tubes (20-24 F) are any better than small tubes (10-16 F) in the management of pneumothoraces. The initial use of large (20-24 F) intercostal tubes is not recommended, although it may become necessary to replace a small chest tube with a larger one if there is a large air leak preventing complete reinflation of the lung [42].

The most common position for chest tube insertion is in the mid-axillary line. This position minimizes the risk of injury to underlying structures, such as the viscera and internal mammary artery. For apical and large pneumothoraces extending to the apex, an antero-apical approach is favored. It requires minimal positioning and rotation of a critically ill patient. The second intercostal space in the mid-clavicular line is often chosen: two fingerbreadths from the lateral sternal border. The internal mammary vessels are at risk and bedside ultrasound may be very helpful in choosing the optimal location while avoiding vascular structures. In patients with persistent air leaks, one-way endobronchial valves are an alternative to surgical intervention [43].

Pleural Effusion

Patients with cancer frequently develop pleural effusions. Of 5888 pleural effusions collected from consecutive patients and analyzed in a single laboratory over a 14-year period, Johnston, et al. reported that 584 (9.9%) samples taken from 472 patients contained malignant cells. Leading etiologies of malignant pleural effusions (MPE) in both sexes were lung cancer (35.6%), lymphoma/leukemia (15.9%), breast cancer (14.8%), female genital cancer (8.1%), and gastrointestinal malignancies (5.9%). The incidence of cancer with unknown primary was 10.2% [44]. Fifteen to 40% of patients with MPE are asymptomatic at the time of diagnosis [45]. On chest imaging, a majority of MPE present as ipsilateral effusions, and approximately 10-13% are bilateral [46]. A quarter of all pleural effusions in a general hospital are due to cancer, and up to 50% of patients with a variety of metastatic malignancies develop a paramalignant or malignant pleural effusion. In addition, 30-70% of all exudative pleural effusions are malignant [47, 48]. Paramalignant pleural effusions are caused by tumor effects on surrounding structures; pleural fluid cytology and pleural biopsy are usually free of cancer cells. Malignant pleural effusions are caused by direct invasion of the pleura, and fluid cytology or pleural biopsies may be positive for malignant cells.

Nearly all malignant or paramalignant pleural effusions become symptomatic. It is advisable to treat these effusions upon presentation, especially if moderate to large in size, since their recurrence may hinder and delay therapy for underlying malignancy. Several important questions face the emergency physician when evaluating patients with malignant pleural effusions. Does the effusion need to be drained? What is the volume of fluid that can be drained safely at one time? What is the appropriate size of the chest tube?

Drainage

Since symptoms are frequent in patients with cancer and pleural effusion, drainage is recommended. For patients who do not require admission after drainage, a simple therapeutic thoracentesis is recommended. We advocate the use of an 8-French pleural drainage catheter, inserted in the posterior axillary line under ultrasound guidance with gravity drainage. We prefer gravity drainage to repeated manual aspirations as it avoids steep fluctuations in negative pleural pressure, results in slower re-expansion of the lung, and allows for pleural pressure monitoring when indicated [49].

Amount of Drainage

Initial thoracentesis is a simple procedure providing symptomatic relief after acute presentation, as well as allowing assessment of the degree of lung re-expansion, which may be important to determine future management strategies. It has been advocated to drain only 1.5 liters during therapeutic thoracentesis, the rationale being to minimize the risk of reexpansion pulmonary edema (REPE) [50]. We believe in draining the majority of the pleural effusion as much as tolerated by the patient for several reasons: One, the incidence of REPE is extremely rare and typically patients develop early mild symptoms (cough and chest pressure) allowing termination of the procedure [50]. Two, in order to assess whether the patient is a candidate for future pleurodesis with a postthoracentesis chest radiograph, one needs to document the juxtaposition of visceral and parietal pleura. Three, draining the majority of pleural effusion provides the optimal and longest symptomatic relief in between recurrences. The common side effects from large-volume thoracentesis are cough and chest discomfort, believed to be secondary to the sharp drop of intrapleural pressure [50]. Therapeutic thoracentesis should be interrupted in case of development of symptoms, as they may precede more severe complications, such as REPE, if drainage continues. Once symptoms develop, patients should limit deep breathing and refrain from talking in long sentences as further stretching the lung may lead to worsening symptoms. These symptoms typically resolve spontaneously. There is evidence that application of noninvasive positive-pressure ventilation lessens the degree of intrapleural pressure drop during large volume thoracentesis [51].

Chest Tube Size

Small-bore chest drains (defined as size < 16 Fr) are now used extensively in MPE drainage and are recommended in major guidelines (e.g., 2010 British Thoracic Society guidelines) [52]. However, the optimal chest tube size for pleurodesis has not been identified. Older case/comparative studies and one RCT suggested that small bore drains are effective for pleurodesis; unexpectedly, a higher complication rate has also been reported. If presenting with recurrent large malignant pleural effusion, we recommend small-bore (8–16 Fr) chest tube insertion in the posterior axillary line, with connection to a Pleur-evac[®] system (Teleflex, Morrisville, NC) without wall suction or gravity drainage bag. We also advocate early pulmonary consultation in order to plan for longterm fluid management strategies. We do not see any advantages to large-bore chest tube insertion, as documented in previous studies [53].

Indwelling pleural catheters (IPCs) should be considered first-line management of MPEs, in conjunction with standard talc pleurodesis. Recognition of the advantages and disadvantages of each approach allows patients to choose an option. It is acknowledged that the use of IPCs without pleurodesis agents can lead to spontaneous pleurodesis [42, 54], allowing removal of the catheter. In patients who desire faster removal of the catheter, a more aggressive drainage regime or instillation of talc via the IPC is a judicious option [55]. Some special situations may face emergency physicians with patients carrying tunneled pleural catheters who present for worsening symptoms due to inability to drain fluid or clogged catheters. Pulmonary consultation is suggested in these situations, as an evaluation of the pleural space with chest radiograph or thoracic ultrasound is required to evaluate for residual pleural fluid. In cases of clogged tunneled pleural catheters with significant residual pleural fluid, intrapleural catheter instillation of alteplase may be indicated to unclog the tube [55].

Summary

Pleural effusion and pneumothorax are common pulmonary manifestations in patients with cancer, most commonly occurring after required interventions treating the primary malignancy. Early recognition and intervention are indicated as delays may interfere with cancer treatment. Small-bore chest tubes are as good as large bores tubes and should be used primarily. Talc pleurodesis and indwelling pleural catheters effectively manage the symptoms of MPE. The use of bedside ultrasound has improved diagnostic accuracy and minimized complications.

Hemoptysis

Etiologies

One of the most common causes of hemoptysis is lung cancer, along with inflammatory and infectious etiologies, such as bronchiectasis, bronchitis, and tuberculosis [56]. During their lifespan, 20% of patients with lung cancer develop hemoptysis. Non-small-cell lung cancer patients have a higher incidence than small cell. Other malignant causes include endobronchial metastatic carcinoma (melanoma, breast, colon, or renal cancer), bronchial carcinoid in young patients, and Kaposi sarcomas in AIDS patients. Hemoptysis can also be chemotherapy induced, caused by necrosis of large tumors or potential medication side effects. With the development of several antiangiogenic agents inhibiting vascular endothelial growth factor for the treatment of advanced lung cancer, more patients are started on those agents as they progress with their disease. One of the reported side effects of these agents, for example, bevacizumab, is hemoptysis. When hemoptysis from bevacizumab occurs, it is often massive. In fact, life-threatening hemoptysis has been reported in up to 9% in patients receiving bevacizumab [57] and most of the hemorrhages were fatal, occurring during the initial cycles of treatment [58]. It is therefore important to recognize and attribute hemoptysis in patients receiving bevacizumab and realize that it is often fatal. Early critical care and interventional pulmonary or radiology consults are crucial in patients receiving bevacizumab presenting with hemoptysis, even if mild.

Massive Hemoptysis

The definition of massive hemoptysis remains poorly defined in the literature because terms used to describe the amount of hemoptysis are inconsistently utilized. Life-threatening or massive hemoptysis is characterized by a larger volume of blood, while clinical instability is considered to be a better indication of poorer outcome. Reported thresholds for massive hemoptysis in the literature range from 100 to 1000 mL in a 24-h period [59]. In our opinion, massive hemoptysis is present when a patient is coughing more than 100 ml/24 h and raises concerns as to airway protection issues or gas exchange impairment. Massive hemoptysis is the cause of death in 3% of patients with lung cancer. Urgent diagnostic and therapeutic interventions are advised in patients with massive hemoptysis. Several initial steps are important upon presentation: localize the source, protect the airway, optimize gas exchange, and then control the bleeding. A multidisciplinary team approach (pulmonology, interventional pulmonology, interventional radiology, and surgery) should be undertaken as soon as the condition is identified [60].

Clinical suspicion, previous history, and available previous imaging are important in localizing or estimating the location of the bleeding. The suspected bleeding site should be placed in a dependent position in order to prevent spillage or formation of large blood clots into the nonbleeding lung. If massive bleeding is associated with significant symptoms, endotracheal intubation should be performed using at least an 8.0 mm endotracheal tube (ETT). A large ETT tube facilitates access for the therapeutic bronchoscope and provides a route for laser coagulation or cryoprobe-assisted removal of large blood clots. Additionally, the tip of the ETT can be pushed deep into the nonbleeding main bronchus to further protect the nonbleeding side from blood contamination. This maneuver obviously requires bronchoscopic guidance. After intubation, mechanical ventilation may be initiated with optimized settings to achieve adequate oxygenation and ventilation. Also, correction of any underlying coagulopathy is warranted.

Bronchoscopy with instillation of epinephrine or ice-cold saline is a temporary measure to potentially control bleeding until laser ablation equipment is available for more permanent control. Use of iced saline has become common practice. Endobronchial epinephrine and norepinephrine are also commonly used; however, recommendations for safe dosage and concentration vary quite widely in the literature. In contrast to iced saline, the safety of endobronchial epinephrine has been questioned, given reports of coronary vasospasm and arrhythmia [61].

Bronchoscopic measures are effective only for bleeding sites within the reach of the bronchoscope. Otherwise, arteriographic localization of a bleeding site and embolization is the method of choice to control bleeding from peripheral sites not easily reached through the airway [62]. Surgical evaluation and possible intervention are indicated in case of an uncontrollable bleeding site [63]. Surgical intervention in bleeding patients is associated with significant morbidity and mortality [64]. Tranexamic acid (TXA), an antifibrinolytic medication that competitively inhibits plasminogen activation, has been prospectively studied in groups of submassive hemoptysis in both IV and nebulized form. TXA was associated with a decrease in hemoptysis and need for interventional procedures [65]. Additionally, recombinant activated factor VII and recombinant thrombin have also been used in cases of hemoptysis due to diffuse alveolar hemorrhage with moderate hemoptysis [66-68]. One-way endobronchial valves have also been used in a small study of patients with moderate hemoptysis [69]. Most available literature on massive hemoptysis is limited.

Summary

Hemoptysis in cancer patients is a serious manifestation of airway involvement or chemotherapy side effect. Hemoptysis is often massive and potentially fatal in patients receiving bevacizumab. Early intervention and consultation of specialized services are crucial steps in the management of massive hemoptysis.

Chemotherapy and Radiation-Related Pulmonary Toxicities

Chemotherapy-Related Pulmonary Toxicity

Acute Complications

Immune Checkpoint Inhibitor-Related Pulmonary Toxicity Immune checkpoint inhibitors (ICIs) have revolutionized cancer treatment and are now a standard of care for a variety of cancers in both metastatic and adjuvant settings. Despite the clinical benefits of the ICI therapy, they are not without complications and can affect multiple organs. ICI-associated pneumonitis is an uncommon, but potentially serious, adverse event.

Lung Toxicity ICI-associated pneumonitis is defined as focal or diffuse immune-related inflammation of lung parenchymal cells and ICI-induced pneumonia is defined as the infection of lungs after ICI treatment. The incidence of pneumonitis is variable in clinical trials, ranging between 3% and 5% [70]. The overall mortality rate associated with all PD-1/ PD-L1 inhibitors is about 0.45% but ICI-induced pneumonitis is the most common reason for death among patients treated with ICIs [71].

The duration from ICI administration to onset of pneumonitis is variable (2-24 months). Early onset of ICI pneumonitis was found to be more severe in one study [72]. No definite clinical symptoms, characteristic CT manifestations, or serological markers are available to establish a diagnosis of ICIinduced pneumonitis. It is usually a diagnosis of exclusion after an extensive evaluation for potential infectious etiologies. Computed tomography (CT) usually displays reticular infiltrates with ground-glass opacities and consolidation in patients receiving PD-1 inhibitors [73]. Supportive care and empiric antibiotics are recommended during the initial treatment. Most ICI-associated pneumonitis cases are corticosteroid therapy responsive; however, 15-30% of cases can show a poor response to therapy, leading to a poor prognosis for these patients. High-dose steroids are advised with a taper over 6 weeks or longer.

Conclusion

Treatment with immune checkpoint inhibitors induces a variety of adverse events, which can sometimes be serious or fatal, albeit infrequent. Among them, pneumonitis is one of the most life-threatening adverse events. Currently, no optimal management guideline exists for the management of pneumonitis, but it usually responds to immunosuppressive medications.

Conventional Chemotherapy-Related Pulmonary Toxicity

Pulmonary toxicity from antineoplastic agents is common. The exact incidence of lung toxicity remains unclear; this is related to confounding factors, including pulmonary comorbidities and prior or concurrent use of other treatment modalities, such as radiation and antineoplastic drugs. It is estimated that approximately 10–20% of patients receiving chemotherapy develop some form of lung toxicity [74, 75].

Pulmonary injury can vary from mild to severe and is divided into acute and delayed onset [76]. The acute onset syndromes include inflammatory interstitial pneumonitis, pulmonary edema, bronchospasm, and pleurisy or pleural effusion, typically presenting after the first one or two cycles of the administered agent.

The inflammatory interstitial pneumonitis syndrome is a hypersensitivity-like reaction. It is the most common chemotherapy-associated lung injury [76]. It has an acute to subacute presentation with productive cough and worsening dyspnea, associated with low-grade fevers and fine crackles on physical exam. A chest radiograph shows either an interstitial or mixed interstitial and alveolar pattern. The most common agents associated with this syndrome are methotrexate, bleomycin, procarbazine, and carmustine. It can mimic an atypical pneumonia or hypersensitivity pneumonitis. The prognosis is generally favorable with simple discontinuation of the offending agent and treatment with corticosteroids for more severe cases.

The pulmonary edema syndrome, less common than interstitial pneumonitis syndrome, has a more acute presentation. Caused by endothelial inflammation and vascular leak, it leads to noncardiogenic pulmonary edema (NCPE). Patients may present with severe dyspnea. Cough, fatigue, and increased work of breathing may also be present, sometimes associated with profound hypoxemia and crackles on physical exam [77]. Chest radiograph shows similar findings to those found in patients with pulmonary edema but with a normal size heart. It is typically a diagnosis of exclusion after establishing normal cardiac function in a clinical picture of congestive heart failure. Chemotherapeutic agents most commonly associated with NCPE are (by order of frequency) cytarabine, interleukin 2, trans-retinoic acids, and gemcitabine. A higher incidence of NCPE has been reported in patients undergoing allogenic and autologous bone marrow transplant, beginning with induction chemotherapy [78, 79].

Bronchospasm and asthmatic-like reactions can also occur with chemotherapy agents. They have been reported upon exposure to the first cycle. Two mechanisms are responsible: IgE and non-IgE related. The chemotherapy agents causing IgE-related bronchospasm are platinum compounds, such as cisplatin, carboplatin, and oxaliplatin. Those causing non-IgE-related bronchospasm are taxanes (paclitaxel and docetaxel), asparaginase, and epipodophyllotoxins (etoposide and teniposide). In the acute setting, anaphylactoid reactions in the absence of hypotension should be treated with intramuscular injections of epinephrine 1:1000, 0.5 mg per single dose. In the presence of severe hypotension or shock, intravenous infusion of epinephrine continuous is recommended. Antihistamines and corticosteroids, as well as bronchodilators and supplemental oxygen, are also indicated.

Pleurisy or pleural effusion can be a manifestation of chemotherapy-induced side effects with methotrexate. When administered in high doses, it may cause chest pain, sometimes 2–5 days later, in 2–4% of patients. Thirty percent of those may progress and develop pleural effusion [80]. Pain typically subsides 3–5 days after discontinuation of the drug and may relapse if the offending agent is restarted [81].

Late Complications

Late-onset chemotherapy-related pulmonary complications usually present 2 months after completion of therapy. The most common manifestation is pulmonary fibrosis. The agents most commonly associated with this complication are bleomycin, busulfan, carmustine, and mitomycin-C. Common risk factors for this toxicity are advanced age, concomitant radiation treatment, or combination chemotherapy. The use of supplemental oxygen, even at low flow rates, amplifies bleomycin toxicity and may play a role in the development of pulmonary fibrosis, even years after treatment [82].

Patients typically present with the insidious onset of dyspnea associated with nonproductive cough. Physical examination reveals crackles and chest radiograph shows bibasilar reticular interstitial markings. Pulmonary function tests may be consistent with restrictive disease. History and physical examination with elimination of other underlying issues, such as congestive heart failure, are essential to the diagnosis. Bronchoscopy may be useful in ruling out lymphangitic spread as well as infectious etiologies. Definitive diagnosis may require video-assisted thoracoscopic surgery. Supportive treatments and a trial of corticosteroids are mainstays of therapy; however, use of oxygen in patients who have bleomycin lung toxicity should only be used in those with severe hypoxemia.

Radiation-Related Pulmonary Toxicity Radiationinduced lung injury (RILI) encompasses any lung toxicity induced by radiation therapy (RT) and manifests acutely as radiation pneumonitis and chronically as radiation pulmonary fibrosis [83]. Radiation-induced lung injury results from the combination of direct radiation cytotoxicity in addition to radiation-induced cellular signal transduction. This cellular activation initiates a repair process that involves cytokines and growth factors, such as basic fibroblast growth factor, interleukin-1, and transforming growth factor-beta, leading to the development of fibrosis. The incidence of serious pulmonary complications from radiation therapy has decreased secondary to advances in radiation delivery techniques. More recent data suggest RILI incidence is highest for lung cancer (5-25%), followed by mediastinal lymphoma (5–10%) and breast cancer (1–5%) [84].

Several risk factors have been described, including volume of lung irradiated, total dose of radiation >60 Gy, number of fractions delivered, concomitant chemotherapy, previous radiation treatment, and weaning of systemic steroids. Age is not a risk factor but radiation pneumonitis seems to be worse in elderly patients. Radiation pneumonitis occurs within 6 months of therapy (most often within 12 weeks), whereas radiation pulmonary fibrosis occurs >1 year following therapy [85]. If symptoms begin earlier, patients tend to suffer a more severe course. The severity of radiation pneumonitis varies from radiographic findings with no clinical symptoms to life-threatening disease requiring hospitalization [86]. The most common symptom is dyspnea, which may be associated with a pinkish productive cough. The diagnosis is usually clinical and based on the timing of radiation treatment and typical chest radiograph findings corresponding to the field of radiation. Clinically significant radiation-induced pulmonary fibrosis typically occurs months to years following therapy and is described as progressive dyspnea associated with lung scarring. Tachypnea and cyanosis are both signs of advanced disease.

Bronchoscopy is rarely helpful and serves only to rule out an infectious or recurrent malignant process. Corticosteroid therapy is commonly used, although its efficacy is controversial. Prednisone (1 mg/kg or equivalent doses of other corticosteroids) is indicated for acute radiation pneumonitis, but not in fibrosis. Therapy should be continued for 2–4 weeks, followed by a slow tapering of the medication for an additional 6–12 weeks [87]. The only two drugs that have shown efficacy in reducing rates of pneumonitis in humans are amifostine and pentoxifylline in combination with tocopherol. Amifostine is the only FDA-approved drug, but its use is limited because of adverse effects [88].

Another reported and more acute form of radiationinduced lung injury is radiation-related bronchiolitis obliterans with organizing pneumonia (BOOP) [89]. Most reported cases occur among patients irradiated for breast cancer. Common manifestations include cough and fever and, to a lesser degree, dyspnea. The radiographic findings begin in the radiation field but may progress even to the contralateral lung in 40% of cases. Patients respond dramatically to corticosteroids but also carry the risk of significant relapse if tapered in a short period of time.

Summary

Radiation pneumonitis often progresses to lung fibrosis and is typically limited to the radiation field. Clinical and radiographic suspicions are important in establishing the diagnosis. Radiation-induced BOOP is a more acute form that often involves the contralateral lung. Treatment with corticosteroids should be tapered slowly in order to avoid the risk of a relapse.

Nonthrombotic Pulmonary Embolism (NTPE)

NTPE is the embolization of nonthrombotic tumor material into the pulmonary circulation, blocking it either entirely or partially. The nonthrombotic tumor material in patients with cancer includes macro- or microembolism. It is called pulmonary tumor embolism (PTE). These emboli are distinct from true metastases as they remain intravascular and rarely invade the pulmonary parenchyma. With complete occlusion, necrosis of the dependent pulmonary parenchyma similar to a thrombotic event follows. When partially occluding the vascular lumen, inflammatory reaction, vascular intimal proliferation, and activation of the coagulation cascade may develop. The reported incidence of this complication is difficult to assess and is reported at 2.4-26% based on postmortem examination [90, 91]. This variability reflects the difficulty in diagnosing this syndrome. There is predominance of digestive system and liver tumors associated with PTE; breast cancer and cardiac lymphomas have also been reported, but at a lesser rate. The risk of tumor embolization is increased among patients undergoing chemotherapy, radiation, or surgical intervention (fragmentation and embolization of tumor fragments or cells). The presentation is often insidious, progressing over several weeks to months. In rare instances, it can be acute (10-20%) [92]. In patients with proximal and large tumor emboli, the presentation can be dramatic and acute with signs of right heart failure. Patients typically present with worsening dyspnea, cough, and increased work of breathing, sometimes associated with ascites and peripheral edema, reflecting increased right heart pressure. The gold standard test is pulmonary artery blood cytology, obtained through a pulmonary artery catheter. Though PTE is not considered to be metastasis, the prognosis is still poor. Treatment is supportive and should be directed to the primary tumor. Chemotherapy does not generally affect the prognosis of patients with PTE unless the primary tumor is very chemotherapy responsive (e.g., trophoblastic or Wilms tumors).

Summary

PTE syndrome is often the result of tumor destruction, whether with medications or surgical intervention. Symptoms are insidious, however, may mimic thrombotic events. Diagnosis is often clinical but occasionally can be made with pulmonary artery blood cytology. Prognosis is generally poor.

References

- Allemani C, Matsuda T, Di Carlo V, Harewood R, Matz M, Nikšić M, et al. CONCORD Working Group. Global surveillance of trends in cancer survival 2000-14 (CONCORD-3): analysis of individual records for 37 513 025 patients diagnosed with one of 18 cancers from 322 population-based registries in 71 countries. Lancet. 2018;391(10125):1023–75.
- Siegel RL, Miller KD, Jemal A. Cancer statistics, 2019. CA Cancer J Clin. 2019;69(1):7–34.
- Azoulay E, Mokart D, Kouatchet A, Demoule A, Lemiale V. Acute respiratory failure in immunocompromised adults. Lancet Respir Med. 2019;7(2):173–86.
- Soares M, Salluh JI, Spector N, Rocco JR. Characteristics and outcomes of cancer patients requiring mechanical ventilatory support for >24 hrs. Crit Care Med. 2005;33(3):520–6.
- Azoulay E, Schlemmer B. Diagnostic strategy in cancer patients with acute respiratory failure. Intensive Care Med. 2006;32(6):808–22.
- Moreau AS, Peyrony O, Lemiale V, Zafrani L, Azoulay E. Acute respiratory failure in patients with hematologic malignancies. Clin Chest Med. 2017;38(2):355–62.
- Varon J, Marik PE. Cardiopulmonary resuscitation in patients with cancer. Am J Hosp Palliat Care. 2007;24(3):224–9.
- ARDS Definition Task Force, Ranieri VM, Rubenfeld GD, Thompson BT, Ferguson ND, Caldwell E, Fan E, et al. Acute respiratory distress syndrome: the Berlin Definition. JAMA. 2012;307(23):2526–33.
- 9. Young AY, Shannon VR. Acute respiratory distress syndrome in cancer patients. Oncol Crit Care. 2019:557–82.
- Cardoso EM, Tomadon A, Lovison K, Schmidt RC, Chung TT, Duarte PA. Acute respiratory distress syndrome in cancer patients: epidemiology, risk factors and outcomes. Eur J Oncol. 2018;23(1):30–40.
- Soubani AO, Shehada E, Chen W, Smith D. The outcome of cancer patients with acute respiratory distress syndrome. J Crit Care. 2014;29(1):183e7–183.e12.
- Azoulay E, Lemiale V, Mourvillier B, Garrouste-Orgeas M, Schwebel C, Ruckly S, et al. OUTCOMEREA Study Group. Management and outcomes of acute respiratory distress syndrome patients with and without comorbid conditions. Intensive Care Med. 2018;44(7):1050–60.
- Mokart D, van Craenenbroeck T, Lambert J, Textoris J, Brun JP, Sannini A, et al. Prognosis of acute respiratory distress syndrome in neutropenic cancer patients. Eur Respir J. 2012;40(1):169–76.
- Wheeler AP, Bernard GR. Acute lung injury and the acute respiratory distress syndrome: a clinical review. Lancet. 2007;369(9572):1553–64.
- Azoulay E, Lemiale V, Mokart D, Pène F, Kouatchet A, Perez P, et al. Acute respiratory distress syndrome in patients with malignancies. Intensive Care Med. 2014;40(8):1106–14.
- 16. Bouteloup M, Perinel S, Bourmaud A, Azoulay E, Mokart D, Darmon M. Outcomes in adult critically ill cancer patients with and without neutropenia: a systematic review and meta-analysis of the Groupe de Recherche en Réanimation Respiratoire du patient d'Onco-Hématologie (GRRR-OH). Oncotarget. 2017;8(1):1860–70.
- de Grooth HJ, Geenen IL, Girbes AR, Vincent JL, Parienti JJ, Oudemans-van Straaten HM. SOFA and mortality endpoints in randomized controlled trials: a systematic review and meta-regression analysis. Crit Care. 2017;21(1):38.

- Bayram B, Sanci E. Invasive mechanical ventilation in the emergency department. Turk J Emerg Med. 2019;19(2):43–52.
- Fuller BM, Mohr NM, Dettmer M, Kennedy S, Cullison K, Bavolek R, et al. Mechanical ventilation and acute lung injury in emergency department patients with severe sepsis and septic shock: an observational study. Acad Emerg Med. 2013;20(7):659–69.
- 20. Frat JP, Ragot S, Girault C, Perbet S, Prat G, Boulain T, et al. REVA network. Effect of non-invasive oxygenation strategies in immunocompromised patients with severe acute respiratory failure: a post-hoc analysis of a randomised trial. Lancet Respir Med. 2016;4(8):646–52.
- Zhan Q, Sun B, Liang L, Yan X, Zhang L, Yang J, et al. Early use of noninvasive positive pressure ventilation for acute lung injury: a multicenter randomized controlled trial. Crit Care Med. 2012;40(2):455–60.
- 22. Walkey AJ, Goligher EC, Del Sorbo L, Hodgson CL, Adhikari NKJ, Wunsch H, et al. Low tidal volume versus non-volume-limited strategies for patients with acute respiratory distress syndrome. A systematic review and meta-analysis. Ann Am Thorac Soc. 2017;14(Suppl 4):S271–9.
- Broccard AF, Hotchkiss JR, Vannay C, Markert M, Sauty A, Feihl F, Schaller MD. Protective effects of hypercapnic acidosis on ventilator-induced lung injury. Am J Respir Crit Care Med. 2001;164(5):802–6.
- Contreras M, Ansari B, Curley G, Higgins BD, Hassett P, O'Toole D, Laffey JG. Hypercapnic acidosis attenuates ventilation-induced lung injury by a nuclear factor-κB-dependent mechanism. Crit Care Med. 2012;40(9):2622–30.
- 25. Tang SE, Wu SY, Chu SJ, Tzeng YS, Peng CK, Lan CC, et al. Pre-treatment with ten-minute carbon dioxide inhalation prevents lipopolysaccharide-induced lung injury in mice via down-regulation of toll-like receptor 4 expression. Int J Mol Sci. 2019;20(24):6293.
- 26. Alhazzani W, Alshahrani M, Jaeschke R, Forel JM, Papazian L, Sevransky J, Meade MO. Neuromuscular blocking agents in acute respiratory distress syndrome: a systematic review and metaanalysis of randomized controlled trials. Crit Care. 2013;17(2):R43.
- Papazian L, Forel JM, Gacouin A, Penot-Ragon C, Perrin G, Loundou A, et al. ACURASYS Study Investigators. Neuromuscular blockers in early acute respiratory distress syndrome. N Engl J Med. 2010;363(12):1107–16.
- National Heart, Lung, and Blood Institute PETAL Clinical Trials Network, Moss M, Huang DT, Brower RG, Ferguson ND, Ginde AA, Gong MN, et al. Early neuromuscular blockade in the acute respiratory distress syndrome. N Engl J Med. 2019;380(21):1997–2008.
- Gallo de Moraes A, El-Yafawi R, Oeckler RA. Early neuromuscular blockade in the acute respiratory distress syndrome (comment). N Engl J Med. 2019;381(8):787.
- de Lassence A, Timsit JF, Tafflet M, Azoulay E, Jamali S, Vincent F, et al. OUTCOMEREA Study Group. Pneumothorax in the intensive care unit: incidence, risk factors, and outcome. Anesthesiology. 2006;104(1):5–13.
- Walkey AJ, Summer R, Ho V, Alkana P. Acute respiratory distress syndrome: epidemiology and management approaches. Clin Epidemiol. 2012;4:159–69.
- Mojoli F, Bouhemad B, Mongodi S, Lichtenstein D. Lung ultrasound for critically ill patients. Am J Respir Crit Care Med. 2019;199(6):701–14.
- 33. Chan SN, Okuno SH, Jatoi A. Causes and outcomes of spontaneous pneumothoraces in solid tumor cancer patients: an update for the medical oncologist. J Thorac Oncol. 2006;1(4):335–8.
- Heerink WJ, de Bock GH, de Jonge GJ, Groen HJ, Vliegenthart R, Oudkerk M. Complication rates of CT-guided transthoracic lung biopsy: meta-analysis. Eur Radiol. 2017;27(1):138–48.

- Helgeson SA, Fritz AV, Tatari MM, Daniels CE, Diaz-Gomez JL. Reducing iatrogenic pneumothoraces: using real-time ultrasound guidance for pleural procedures. Crit Care Med. 2019;47(7):903–9.
- 36. Choi CM, Um SW, Yoo CG, Kim YW, Han SK, Shim YS, Lee CT. Incidence and risk factors of delayed pneumothorax after transthoracic needle biopsy of the lung. Chest. 2004;126(5):1516–21.
- Picchi SG, Lassandro G, Bianco A, Coppola A, Ierardi AM, Rossi UG, Lassandro F. RFA of primary and metastatic lung tumors: long-term results. Med Oncol. 2020;37(5):35.
- Moussa AM, Ziv E, Solomon SB, Camacho JC. Microwave ablation in primary lung malignancies. Semin Intervent Radiol. 2019;36(4):326–33.
- Smith SL, Jennings PE. Lung radiofrequency and microwave ablation: a review of indications, techniques and post-procedural imaging appearances. Br J Radiol. 2015;88(1046):20140598.
- 40. Carson-Chahhoud KV, Wakai A, van Agteren JE, Smith BJ, McCabe G, Brinn MP, O'Sullivan R. Simple aspiration versus intercostal tube drainage for primary spontaneous pneumothorax in adults. Cochrane Database Syst Rev. 2017;9(9):CD004479.
- Thelle A, Gjerdevik M, SueChu M, Hagen OM, Bakke P. Randomised comparison of needle aspiration and chest tube drainage in spontaneous pneumothorax. Eur Respir J. 2017;49(4):1601296.
- 42. Bibby AC, Dorn P, Psallidas I, Porcel JM, Janssen J, Froudarakis M, et al. ERS/EACTS statement on the management of malignant pleural effusions. Eur J Cardiothorac Surg. 2019;55(1):116–32.
- Dugan KC, Laxmanan B, Murgu S, Hogarth DK. Management of persistent air leaks. Chest. 2017;152(2):417–23.
- 44. Johnston WW. The malignant pleural effusion. A review of cytopathologic diagnoses of 584 specimens from 472 consecutive patients. Cancer. 1985;56(4):905–9.
- 45. Feller-Kopman DJ, Reddy CB, DeCamp MM, Diekemper RL, Gould MK, Henry T, et al. Management of malignant pleural effusions. An official ATS/STS/STR clinical practice guideline. Am J Respir Crit Care Med. 2018;198(7):839–49.
- 46. Dixit R, Agarwal KC, Gokhroo A, Patil CB, Meena M, Shah NS, Arora P. Diagnosis and management options in malignant pleural effusions. Lung India. 2017;34(2):160–6.
- Ferreiro L, Suarez-Antelo J, Valdes L. Pleural procedures in the management of malignant effusions. Ann Thorac Med. 2017;12(1):3–10.
- 48. Agrawal A, Tandon R, Singh L, Chawla A. Clinico- pathological profile and course of malignant pleural effusion in a tertiary care teaching hospital in western U.P. with special reference to lung cancer. Lung India. 2015;32(4):326–30.
- 49. Lentz RJ, Shojaee S, Grosu HB, Rickman OB, Roller L, Pannu JK, et al. Interventional Pulmonary Outcomes Group. The impact of gravity vs suction-driven therapeutic thoracentesis on pressure-related complications: the GRAVITAS Multicenter Randomized Controlled Trial. Chest. 2020;157(3):702–11.
- Feller-Kopman D, Berkowitz D, Boiselle P, Ernst A. Large-volume thoracentesis and the risk of reexpansion pulmonary edema. Ann Thorac Surg. 2007;84(5):1656–61.
- Abouzgheib W, Arya R, Cruz-Morel K, Koleman D, Kass J, Boujaoude Z, et al. The impact of continuous positive airway pressure upon pleural fluid pressures during thoracentesis. Respiration. 2019;98(1):55–9.
- 52. Roberts ME, Neville E, Berrisford RG, Antunes G, Ali NJ, BTS Pleural Disease Guideline Group. Management of a malignant pleural effusion: British Thoracic Society pleural disease guideline 2010. Thorax. 2010;65(Suppl 2):ii32–40.
- Fysh ET, Smith NA, Lee YC. Optimal chest drain size: the rise of the small-bore pleural catheter. Semin Respir Crit Care Med. 2010;31(6):760–8.
- 54. Wahidi MM, Reddy C, Yarmus L, Feller-Kopman D, Musani A, Shepherd RW, et al. Randomized trial of pleural fluid drainage frequency in patients with malignant pleural effusions. The ASAP trial. Am J Respir Crit Care Med. 2017;195(8):1050–7.
- Syer T, Walker S, Maskell N. The use of indwelling pleural catheters for the treatment of malignant pleural effusions. Expert Rev Respir Med. 2019;13(7):659–64.
- Johnstone C, Rich SE. Bleeding in cancer patients and its treatment: a review. Ann Palliat Med. 2018;7(2):265–73.
- 57. Cho YJ, Murgu SD, Colt HG. Bronchoscopy for bevacizumabrelated hemoptysis. Lung Cancer. 2007;56(3):465–8.
- 58. Han S, Hong Y, Liu T, Wu N, Ye Z. The efficacy and safety of paclitaxel and carboplatin with versus without bevacizumab in patients with non-small-cell lung cancer: a systematic review and metaanalysis. Oncotarget. 2018;9(18):14619–29.
- 59. Fartoukh M, Khoshnood B, Parrot A, Khalil A, Carette MF, Stoclin A, et al. Early prediction of in-hospital mortality of patients with hemoptysis: an approach to defining severe hemoptysis. Respiration. 2012;83(2):106–14.
- Radchenko C, Alraiyes AH, Shojaee S. A systematic approach to the management of massive hemoptysis. J Thorac Dis. 2017;9(Suppl 10):S1069–86.
- 61. Khoo KL, Lee P, Mehta AC. Endobronchial epinephrine: confusion is in the air. Am J Respir Crit Care Med. 2013;187(10):1137–8.
- Panda A, Bhalla AS, Goyal A. Bronchial artery embolization in hemoptysis: a systematic review. Diagn Interv Radiol. 2017;23(4):307–17.
- Andréjak C, Parrot A, Bazelly B, Ancel PY, Djibré M, Khalil A, et al. Surgical lung resection for severe hemoptysis. Ann Thorac Surg. 2009;88(5):1556–65.
- Davidson K, Shojaee S. Managing massive hemoptysis. Chest. 2020;157(1):77–88.
- 65. Wand O, Guber E, Guber A, Epstein Shochet G, Israeli-Shani L, Shitrit D. Inhaled tranexamic acid for hemoptysis treatment: a randomized controlled trial. Chest. 2018;154(6):1379–84.
- Park JA. Diffuse alveolar hemorrhage and recombinant factor VIIa treatment in pediatric patients. Korean J Pediatr. 2016;59(3):105–13.
- 67. Diaz R, Almeida P, Alvarez M, Ferrer G, Hernandez F. Lifethreatening pulmonary hemorrhage responds to recombinant factor VIIa: a case series in South Florida hospitals. Cureus. 2019;11(11):e6202.
- 68. Lee J, Rhee CK, Kim SC, Kim YK, Kim HJ, Lee S, et al. Use of intrapulmonary administration of thrombin in hematological malignancy patients with alveolar haemorrhage: a case series. Medicine (Baltimore). 2020;99(20):e20284.
- 69. Patel B, Abi-Fadel D, Rosenheck J, Bartter T, Boujaoude Z, Abouzgheib W. Endobronchial valves for treatment of hemoptysis. J Bronchol Intervent Pulmonol. 2019;26(2):e22–4.
- 70. Wang Y, Zhou S, Yang F, Qi X, Wang X, Guan X, et al. Treatmentrelated adverse events of pd-1 and pd-11 inhibitors in clinical trials: a systematic review and meta-analysis. JAMA Oncol. 2019;5(7):1008–19.
- 71. De Velasco G, Je Y, Bossé D, Awad MM, Ott PA, Moreira RB, et al. Comprehensive meta-analysis of key immune-related adverse events from CTLA-4 and PD-1/PD-L1 inhibitors in cancer patients. Cancer Immunol Res. 2017;5(4):312–18. Erratum in: Cancer Immunol Res. 2018;6(4):498–9.
- 72. Nishino M, Ramaiya NH, Awad MM, Sholl LM, Maattala JA, Taibi M, et al. PD-1 inhibitor-related pneumonitis in advanced cancer patients: radiographic patterns and clinical course. Clin Cancer Res. 2016;22(24):6051–60.
- Nishino M, Sholl LM, Hodi FS, Hatabu H, Ramaiya NH. Anti-PD-1-related pneumonitis during cancer immunotherapy. N Engl J Med. 2015;373(3):288–90.

- Dimopoulou I, Bamias A, Lyberopoulos P, Dimopoulos MA. Pulmonary toxicity from novel antineoplastic agents. Ann Oncol. 2006;17(3):372–9.
- Vahid B, Marik PE. Pulmonary complications of novel antineoplastic agents for solid tumors. Chest. 2008;133(2):528–38.
- Abid SH, Malhotra V, Perry MC. Radiation-induced and chemotherapy-induced pulmonary injury. Curr Opin Oncol. 2001;13(4):242–8.
- Briasoulis E, Pavlidis N. Noncardiogenic pulmonary edema: an unusual and serious complication of anticancer therapy. Oncologist. 2001;6(2):153–61.
- Cahill RA, Spitzer TR, Mazumder A. Marrow engraftment and clinical manifestations of capillary leak syndrome. Bone Marrow Transplant. 1996;18(1):177–84.
- Bhalla KS, Wilczynski SW, Abushamaa AM, Petros WP, McDonald CS, Loftis JS, et al. Pulmonary toxicity of induction chemotherapy prior to standard or high-dose chemotherapy with autologous hematopoietic support. Am J Respir Crit Care Med. 2000;161(1):17–25.
- Cudzilo C, Aragaki A, Guitron J, Benzaquen S. Methotrexateinduced pleuropericarditis and eosinophilic pleural effusion. J Bronchol Intervent Pulmonol. 2014;21(1):90–2.
- Wang W, Zhou H, Liu L. Side effects of methotrexate therapy for rheumatoid arthritis: a systematic review. Eur J Med Chem. 2018;158:502–16.
- Azambuja E, Fleck JF, Batista RG, Menna Barreto SS. Bleomycin lung toxicity: who are the patients with increased risk? Pulm Pharmacol Ther. 2005;18(5):363–6.
- Hanania AN, Mainwaring W, Ghebre YT, Hanania NA, Ludwig M. Radiation-induced lung injury: assessment and management. Chest. 2019;156(1):150–62.
- Ebert N, Baumann M, Troost EG. Radiation-induced lung damage – clinical risk profiles and predictive imaging on their way to risk-adapted individualized treatment planning? Radiother Oncol. 2015;117(1):1–3.
- Bernchou U, Schytte T, Bertelsen A, Bentzen SM, Hansen O, Brink C. Time evolution of regional CT density changes in normal lung after IMRT for NSCLC. Radiother Oncol. 2013;109(1):89–94.
- Graves PR, Siddiqui F, Anscher MS, Movsas B. Radiation pulmonary toxicity: from mechanisms to management. Semin Radiat Oncol. 2010;20(3):201–7.
- Bledsoe TJ, Nath SK, Decker RH. Radiation pneumonitis. Clin Chest Med. 2017;38(2):201–8.
- Antonadou D, Coliarakis N, Synodinou M, Athanassiou H, Kouveli A, Verigos C, et al. Clinical Radiation Oncololgy Hellenic Group. Randomized phase III trial of radiation treatment +/– amifostine in patients with advanced-stage lung cancer. Int J Radiat Oncol Biol Phys. 2001;51(4):915–22. Erratum in: Int J Radiat Oncol Biol Phys. 2002;52(5):1458. Erratum in: Int J Radiat Oncol Biol Phys. 2002;54(1):308.
- Ducray J, Vignot S, Lacout A, Pougnet I, Marcy PY, Chapellier C, et al. Radiation-induced bronchiolitis obliterans with organizing pneumonia. Cancer Radiother. 2017;21(2):148–54. (Article in French).
- 90. Sakuma M, Fukui S, Nakamura M, Takahashi T, Kitamukai O, Yazu T, et al. Cancer and pulmonary embolism: thrombotic embolism, tumor embolism, and tumor invasion into a large vein. Circ J. 2006;70(6):744–9.
- Ma SQ, Lin Y, Ying HY, Shao YJ, Li XY, Bai CM. Solid malignancies complicated with pulmonary embolism: clinical analysis of 120 patients. Chin Med J. 2010;123(1):29–33.
- Veinot JP, Ford SE, Price RG. Subacute cor pulmonale due to tumor embolization. Arch Pathol Lab Med. 1992;116(2):131–4.

Venous Thromboembolism

Nicholas R. Pettit and Jeffrey A. Kline

Case Study

A 55-year-old male presents to the emergency department (ED) with chest pain. It started about 1 day ago, constant in nature, substernal, nonexertional, and pleuritic. He took some acetaminophen without any relief. He has had some dyspnea on exertion for the past week, but no fever, chills, myalgias, leg pain, or swelling. He has had some anorexia because he is undergoing chemotherapy, and last received gemcitabine 2 weeks ago. His vital signs are blood pressure of 145/85, heart rate of 75 beats per minute, SpO₂ 99% on room air, and a temperature of 37.4 °C. He is comfortable in the bed and otherwise is well appearing in no acute distress. You suspect pulmonary embolism, order troponin and pro-BNP which return normal, and the ECG is sinus rhythm unchanged from his previous. Remainder of labs including BMP and CBC are within normal limits. CT scan of his chest demonstrates several small subsegmental pulmonary emboli without any evidence of right heard strain. As the treating practitioner in the ED, you are responsible for this patient's disposition, and ultimately decide to discharge this patient after communication with the patient's oncologist. Application of the Hestia criteria determines that the patient is low risk for outpatient management of his pulmonary emboli, so you prescribe rivaroxaban and discharge the patient home after arranging follow-up with his oncologist in 2 days.

J. A. Kline

Pathophysiology of Thromboembolism in Cancer

Annually, 1.5 million patients will receive a new diagnosis of cancer in the United States, of whom 5% or 75,000 patients will receive an additional diagnosis of venous thromboembolic disease (VTE). VTE is a direct complication of cancer and is known to be influenced by tumor type, stage, and active treatment [1]. The risk of VTE is 53-fold higher than baseline in the first 3 months after diagnosis of cancer and remains approximately fourfold higher until 15 years after initial cancer diagnosis [2]. Carrying a dual diagnosis of both cancer and pulmonary embolism (PE) is associated with worsened prognosis, increased recurrence rates, longer duration of anticoagulation, and worsened bleeding risks.

In healthy patients, the vessel endothelium prevents thrombus formation by acting as a barrier to the underlying subendothelium, which contains tissue factor (TF). When TF is exposed through vessel wall damage, it complexes with circulating Factor VIIa and initiates the extrinsic pathway, the primary driver behind the entire coagulation cascade. To initiate a clot, TF must bind with circulating Factor VIIa. This results in a complex capable of converting Factor X to Xa, which cleaves prothrombin to thrombin, in turn cleaving fibrinogen to fibrin, leading to a cross-linked fibrin clot after the action of Factor 13, itself activated by thrombin. Cancer increases the exposure of TF to the blood by several mechanisms, including the surface characteristics of cancer cells, their production of TF-bearing microparticles, and by direct vascular damage as a result of tumor spread. As an example, pancreatic cancer, a highly thrombogenic malignancy, causes significant elevations in microparticle-associated TF, leading to rates of VTE as high as 45% in some series [3, 4]. In addition to increased production of TF, cancers can also release various proinflammatory cytokines, interleukins, and procoagulants [5].

Cancer patients undergo a variety of procedures and treatments that further increase their risk for thrombosis. Surgery, chemotherapy, and radiation therapy cause a proinflammatory

Check for updates

N. R. Pettit (🖂)

Department of Emergency Medicine, Eskenazi Hospital, Indiana University School of Medicine, Indianapolis, IN, USA e-mail: nrpettit@iu.edu

Department of Emergency Medicine, Eskenazi and Methodist Hospitals, Indiana University School of Medicine, Indianapolis, IN, USA

state, and some chemotherapeutic agents produce a prothrombotic state that independently increases clotting risk, notably fluoropyrimidines, L-asparaginase, and thalidomide derivatives [6, 7]. The chemotherapeutic agents Ara C and 5-fluorocytosine alter the metabolism of coumarins and complicate the ability to achieve stable anticoagulation. In addition, interrupting anticoagulation for procedures in patients with active VTE has been associated with increasing rates of postoperative thrombotic and bleeding complications [8]. Cancer patients frequently have other risk factors for VTE including indwelling catheters, immobility, and folate deficiency [9].

Thrombogenic Cancers

Emergency physicians will often consider the need to test for VTE while treating patients with cancer. However, thrombogenicity varies with host factors, tumor stage, and type. In general, the more undifferentiated the cell type and the larger the tumor burden, the higher the risk. The incidence of VTE in a large study population demonstrated a rate of 2% of all cancer patients, with increased rates seen among patients with metastatic disease, pancreatic cancer, and colon cancer [10]. Cancers that appear to confer no or minimal risk of VTE include localized breast, cervical, prostate, and nonmelanomatous skin cancers, such as squamous cell carcinoma and basal cell carcinoma. However, advanced stage breast cancer or breast cancer patients undergoing chemotherapy may have a thrombosis risk of up to 10% during treatment [11, 12]. Similarly, treatment of leukemias, particularly acute lymphocytic leukemia treated with L-asparaginase and acute promyelocytic leukemias treated with all-trans-retinoic acid have each been associated with an approximate 10% incidence of VTE throughout the course of therapy. Other cancers that are particularly thrombogenic include adenocarcinoma, glioblastoma, melanoma (in contrast to other skin cancers), lymphoma, and multiple myeloma [13]. Pancreatic, gastric, ovarian, and renal cell cancers carry notoriously high risks. Finally, in addition to different risks based on malignancy types, within the California Cancer Registry (CCR), racial disparities exist with non-Hispanic whites and African Americans having an incidence of VTE almost twice that of Hispanic and Asian patients. This may be due in part to variable expression patterns in different ethnic groups, but the basis for these disparities are mostly undefined [14, 15].

Clinicians should be especially vigilant for VTE during the induction phase of chemotherapy, as this is the most thrombogenic period [16]. L-asparaginase and bolus fluorouracil treatment confer particularly high thrombosis risks, probably by reducing antithrombin concentrations [17]. While localized breast cancer has a relatively low thrombogenic potential, risk approximately doubles with tamoxifen
 Table 24.1
 Diagnostic criteria required to exclude venous thromboembolism in cancer patients

Deep vein thrombosis	Pulmonary embolism
Negative full leg duplex ultrasonography ^a	Adequate quality negative CTPA
Negative (<500 ng/µL) d-dimer plus negative proximal ultrasound	Negative homogenous perfusion scan
Two negative proximal ultrasounds 2–7 days apart ^b	Low probability V/Q scan and a single negative bilateral whole-leg lower extremity duplex ultrasonography or two negative proximal ultrasounds 2–7 days apart

CTPA Computed tomography pulmonary angiography, V/Q A ventilation–perfusion scan

^aFull leg ultrasound includes spectral and B-mode compression imaging of the proximal and distal femoral vein, the popliteal, posterior tibial, peroneal and greater saphenous veins

^bProximal ultrasound includes spectral and B-mode compression imaging of the proximal and distal femoral vein and the popliteal vein

treatment, whereas aromatase inhibitors do not appear to increase risk. Concomitant treatment with red cell growth factors such as erythropoietin clearly increases risk of thrombosis, regardless of tumor type or stage [18]. Any patient who presents with extremity swelling or chest pain during the initial treatment phase with these drugs should undergo diagnostic studies outlined in Table 24.1 to exclude thromboembolism. Similarly, multiple myeloma patients treated with lenalidomide or thalidomide are at risk, although one large Japanese cohort study found this to be 1.4% and at baseline for their cancer [19, 20]. Bevacizumab presents a complex picture, with some studies suggesting a high risk and a more recent systematic review showing no increase compared with matched patients receiving other forms of chemotherapy for similar tumors [21]. A clinical prediction rule has been developed [18] to determine which cancer patients are at highest risk for future thrombosis. In this model, patients with \geq 3 points are found to have VTE risks of approximately 7%. Lastly, a recent systematic review and meta-analysis reports a pooled VTE rate of approximately 7% among cancer patients receiving neoadjuvant therapy, with highest rates among those with bladder and esophageal cancers [22].

What Does VTE Mean for the Cancer Patient?

Cancer patients who develop VTE have higher morbidity and mortality risks. This risk represents the synergistic effect between the two disease entities. Diagnosis of VTE in cancer confers several independent, negative consequences, including a reduced overall probability of survival, indication of more aggressive cancer, and a higher risk of bleeding from anticoagulation than noncancer patients that suffer from VTE [23]. In addition, cancer-associated nausea impairs oral anticoagulant administration and absorption. Pill fatigue can impair anticoagulation therapy compliance, as these patients are often on multiple medications. For those receiving vitamin K antagonists (VKAs) such as warfarin, the additional needle sticks for INR checks can be onerous. Finally, cancer patients tend to experience higher severity of clot burden than patients without cancer. In the authors' experience, these thrombi are larger and more extensive in both the extremities and in the lung, leading to a larger disease burden and increased incidence of postthrombotic syndrome.

Special Considerations for Diagnosis

The diagnostic approach to a cancer patient with suspected VTE mirrors that of other moderate- to high-risk patients with suspected PE (see Table 24.1). Unfortunately, while being at higher risk for thrombosis, cancer patients are also at higher risk for complications from diagnosis. One such currently debated topic is that of contrast-induced nephropathy (CIN), as older literature illustrates an increased risk of CIN in all outpatients receiving contrast-enhanced computed tomography, as well as in active cancer patients receiving contrast-enhanced CTs [24, 25]. More recently, however, in a retrospective cohort comparing contrasted to noncontrasted CT scans in a cancer cohort, the rate of acute kidney injury (AKI) was independent of the administration of contrast, but instead was more associated with preexisting congestive heart failure and prior AKIs [26].

The D-dimer, which normally can be used to reliably exclude VTE because of its high sensitivity, becomes markedly less specific in the cancer population. Many cancer patients will have positive D-dimer results in the absence of clot, meaning that the test is less useful in patients with active cancer. Active malignancy induces a system-wide activation of the clotting network, and elevated D-dimer levels may herald a poor cancer prognosis without indicating thrombosis [27]. This phenomenon likely reflects important interactions between hemostatic mechanisms and tumor-associated angiogenesis and inflammatory mediators. D-dimer levels confer additional prognostic value to tumor stage and predict decreased survival for breast cancer patients, and is a marker of decreased survival in colorectal, cervical, pancreas, prostate, brain, and melanoma cancers [1]. While a negative D-dimer still helps rule out VTE, it does so in a smaller proportion of patients with active cancer.

Risk Stratification and Management

In the era of target-specific anticoagulants (TSAs, formerly referred to as direct oral anticoagulants or DOACs) such as rivaroxaban, apixaban, and dabigatran, the need to hospitalize all patients with thromboembolic event, including PE, is being called into question. It has been long held that many

patients with DVT can be safely discharged home [28]. However, several caveats to this practice exist, chief among them is a low-risk status. In practice, low-risk status among those with cancer refers to the lack of a reason why the patient would need hospital care in the next 30 days. The primary determinant of success of outpatient treatment will be access to anticoagulant and ability to administer it. While the preferred method of treatment for both DVT and PE in active cancer remained injected low-molecular-weight heparin (LMWH), rivaroxaban is now an acceptable primary therapy for VTE according to The National Comprehensive Cancer Network Guidelines [29, 30]. Thus, the first objective with home treatment is to determine if the patient and caretakers have the capacity and competence to administer twice daily subcutaneous injections. Other predictors of return to the hospital are inadequate pain control, decompensation of other disease processes, need for oxygen, or hemodynamic and respiratory effects of concomitant PE, which occur in over 1/3 of patients [31]. Patients with significant iliofemoral clot burden may require admission for catheter-based therapy in view of evidence that this approach significantly decreases postthrombotic syndrome and leg ulceration, which are the major complications of these proximal, occlusive DVTs [32].

Patients with PE are increasingly treated at home, starting the same day of diagnosis, provided that these patients meet low-risk criteria [32–34]. Several criteria have been validated, including the Hestia criteria, the PESI, and sPESI scores [35–38]. However, the sPESI score excludes discharge of patients with cancer. The Hestia criteria do not exclude patients with cancer, but we suggest that cancer patients with VTE must also meet a separate set of rules, of which two have currently been published: POMPE-C and criteria derived from the Registro Informazitado Enfermo TromboEmbolica (RIETE) database by den Exter et al. [39]. These criteria include additional predictors such as metastases, immobilization, and low body weight while they also stratify cancer patients based on perceived risk of thrombosis [39, 40].

Thus, to deem a patient with either DVT or PE and with active cancer (defined as under the current care of an oncologist or receiving palliative therapy, or any patient with metastasis) as low risk for home treatment, we recommend a two-step approach. First, apply the Hestia criteria (Table 24.2) and second, for patients with confirmed PE or patients with DVT and strongly suspected PE, apply either the POMPE-C criteria (Table 24.3) or the den Exter criteria [35, 41–43]. The POMPE-C criteria predicts mortality for cancer patients with PE; however, for safe outpatient management, the Hestia criteria account for several other variables that may particularly impact cancer patients (e.g., renal dysfunction, liver disease, medical/social reasons for admission). The EPIPHANY index has been derived and validated for pre-

Table 24.2 The Hestia criteria

Patient fails criteria if any of the below are true
Hemodynamically unstable
Requires thrombolysis or embolectomy
Active bleeding or high risk for bleeding
More than 24 h supplemental O_2 required to maintain O_2 saturation >90%
PE diagnosed while currently under active anticoagulant therapy
Severe pain requiring >24 h intravenous analgesic therapy
Medical or social reason for hospitalization >24 h (infection, malignancy, no support system, etc.)
Creatinine clearance <30 mL/min
Severe liver impairment
Pregnancy
Documented history of heparin-induced thrombocytopenia

 Table 24.3
 The POMPE-C criteria (used in an online calculator for overall risk of death at 30 days)

If all criteria shown are absent, the patient's risk of death is sufficiently low to justify home treatment for the patient with active cancer Absence of a *Do Not Resuscitate* order No respiratory distress (defined by the patient showing fear, anxiety, or dyspnea) No unilateral leg swelling No altered mental status Heart rate <100 beats/min Respiratory rate <28 breaths/min Pulse oximetry >94% on room air Weight >140 pounds

dicting risk of serious complications from incidental PEs in cancer patients but has not been used to demonstrate safe ED-discharge among acute DVT/PE in cancer patients [44].

Treatment

At present, the recommended treatment for VTE in cancer is daily injections of 200 IU/kg bodyweight of dalteparin, a LMWH, as opposed to warfarin. The CLOT trial in 2003 demonstrated that dalteparin was superior to warfarin with a 52% risk reduction profile for recurrence of clot [45]. It is important to recognize that there is a widespread assumption that enoxaparin 1 mg/kg bodyweight (Lovenox[®]) is equivalent to dalteparin (Fragmin[®]), although this has never been demonstrated in a clinical trial.

Despite the knowledge of a risk reduction from dalteparin, patients often request oral therapy, as injections are expensive, onerous, and painful, as well as causing bruises and disfigurement [37]. While limited data suggest that patients prefer efficacy and safety over convenience [46], it is the authors' opinion that physicians underestimate the negative perceptions that patients have toward injections. Given the difficulty of self-injection and cost of the injectable LMWHs, many EDs utilize case managers or social workers to assist with the transition to outpatient therapy.

FDA-approved drugs, specifically target-specific anticoagulants (TSA), include apixaban (Eliquis[®]), betrixaban (Bevyxxa[®]), dabigatran (Pradaxa[®]), edoxaban (Savaysa[®]), and rivaroxaban (Xarelto[®]). These drugs offer alternative to therapies more inconvenient injectable LMWHs and VKAs as they are easier to take and less invasive for patients [47]. Furthermore, they don't require weight-based dosing, facilitating outpatient compliance and management.

Available data support a shift toward the use of orally available TSAs for treatment of patients with active cancer and VTE. The basis for this statement comes from a pooled subgroup analysis of patients with cancer at the time of enrollment in the EINSTEIN DVT and PE studies [38]. Together, these studies randomized 597 patients with either active cancer at baseline (n = 430) or cancer diagnosed during the study (n = 167) and found statistically insignificant reductions in risk of both VTE and bleeding for rivaroxaban versus standard therapy (LMWH + VKA) while producing a net clinical benefit for the composite outcome of recurrent VTE and major bleeding (hazard ratio, 0.60; 95% CI, 0.36-0.99) favoring rivaroxaban. This is reinforced by a Cochrane review suggesting that LMWHs compared to VKAs produce a reduction in VTE and in comparing TSAs to LMWHs finding a reduction in VTE but an increased risk of major bleeding [48]. Results from a trial dedicated to treating patients with VTE and active cancer suggest equal efficacy of rivaroxaban to LMWH but with a slightly higher rate of clinically relevant bleeding [49]. Lastly, comparing oral rivaroxaban to LMWH and VKA in the treatment of symptomatic VTE in cancer patients, rivaroxaban had similar efficacy and reduced number of major bleeding events as compared to LMWH/ VKA but had no difference in clinically relevant bleeding [50]. It is reasonable to assume equipoise between injectable LMWH and the direct-acting anticoagulants. Figure 24.1 highlights a summary of randomized trials on TSAs in patients with cancer, both in terms of completed work and ongoing trials [51].

Deciding on which therapy to initiate in the ED should include patient preferences that might influence adherence (some patients hate injections; others with nausea may want to avoid swallowing pills) [48]. The Anti-Clot Treatment Scale is a 15-item patient report scale developed and validated for use in measuring patient-reported satisfaction with anticoagulation treatment [52]. This scale has been utilized in several trials to demonstrate that treatment with oral TSAs results in improved treatment satisfaction as compared to enoxaparin/VKA, as well as improved psychological impacts as compared to VKAs [53, 54]. Similarly, in a systemic review of patient-reported outcomes associated with TSAs, patients prefer TSAs to VKA with higher quality of life scores and increased adherence

Randomized trials in patients with cancer



Fig. 24.1 Summary of randomized trials on target-specific anticoagulants in patients with cancer. (From Vedovati et al. [51], with permission from Elsevier)

[55]. Lastly, twice daily dosing with LMWH in cancer patients was nonsignificantly associated with a trend toward worsening perceptions of the 6-month VTE LMHW treatment, further suggesting that daily dosing is preferred, or additionally suggesting that TSAs may be more tolerable in the long term [56].

Duration of need for anticoagulation sometimes emerges as a concern in the ED, particularly for patients who present with bleeding. Although several stopping criteria have been derived [57], none are adequately validated to provide clear, binary decision-making for cancer patients. A general rule of thumb is to anticoagulate for the duration of cancer treatment and then for several months thereafter. At a minimum, cancer patients with any venous thrombosis, even if distal or superficial, should be treated with anticoagulants for 3 months and patients with proximal DVT or any PE for 6 months [57]. There is some evidence that men who develop PE during cancer should remain on lifelong anticoagulation. Thus, the prudent emergency practitioner would be wise to consider thrombosis as a cause for disease in patients recently in remission. Patients with a history of cancer that is inactive and who develop thrombosis should be treated in accordance with guidelines recommended for a patient with unprovoked DVT (3-6 months) and PE (minimum of 6 months).

 Table 24.4
 Risk tool of Khorana for prediction of which patients undergoing chemotherapy will develop venous thromboembolism

Site of cancer	Risk score
Very high risk (stomach, pancreas)	2
High risk (lung, lymphoma, gynecological, bladder, testicular)	1
Prechemotherapy platelet count >350,000/µL	1
Hemoglobin <10 g/dL or red cell growth factors	1
Prechemotherapy white blood count >11,000/µL	1
Body mass index >35 kg/m ²	1

Thromboembolism Prophylaxis

Some malignancies confer an extraordinarily high risk of VTE, prompting the consideration to prescribe a prophylactic dose of anticoagulant [14]. Rivaroxaban was utilized in a double-blind, randomized trial of 841 high-risk, ambulatory, cancer patients, which demonstrated rivaroxaban did not significantly reduce the incidence of VTE or death from VTE over 180 days [58]. Conversely, several other studies have demonstrated some utility in VTE prophylaxis, such as in a study with 563 intermediate-to-high risk for VTE cancer patients (Khorana score ≥ 2 , Table 24.4), using apixaban for VTE prophylaxis during chemotherapy. Primary outcomes were incidence of VTE in 180 days and major bleeding, and

 Table 24.5
 Contraindications to anticoagulation in the cancer patient

 [62]
 [62]

Absolute contraindications
Active, major, serious, or potentially life-threatening bleeding that is
not immediately reversible with medical or surgical therapies
Malignant hypertension, severe and uncontrolled
Liver failure leading to uncompensated coagulopathy (cirrhosis)
Severe platelet dysfunction, known bleeding diathesis, inherited bleeding disorder
Persistent thrombocytopenia (<20,000/µL)
High-risk invasive procedure needed (lumbar puncture)
TSA specific: concurrent use of P-glycoprotein or CYP3A4 inhibitor
Relative contraindications
Intracranial or spinal lesion at risk for bleeding
Active gastrointestinal bleed or risk for bleeding (ulceration)
Active but non-life-threatening bleeding (hematuria)
CNS bleeding in past 4 weeks
Recent high-risk procedure/intervention
Persistent thrombocytopenia (<50,000/µL)
Consider not starting anticoagulation
End of life, hospice patient
Asymptomatic thrombosis at high risk for bleeding
Patient refusal or against their wishes
Risk for nonadherence

apixaban was found to significantly reduce the rate of VTE compared to placebo, however, at an expense of increased major bleeding [59]. Similar other trials utilizing nadroparin and semuloparin have demonstrated reduction in thromboembolic events in cancer patients actively receiving chemotherapy [60, 61].

Given numerous recent studies exploring the efficacy and utility of VTE prophylaxis, a meta-analysis was completed of randomized controlled trials published over a 4-year period (2014–2018) [62]. The American Society of Clinical Oncologic (ASCO) recommends thromboprophylaxis with apixaban, rivaroxaban or LMWH for high-risk outpatients with cancer. Rivaroxaban and edoxaban have added as options for VTE treatment. Lastly, other recommendations have been made regarding VTE treatment in the setting of brain metastases, hospitalization prophylaxis, and prophylaxis in the setting of major cancer surgery. Concurrently, the emergency medicine physician must be aware of contraindications to therapy and these are listed in Table 24.5 [62]. Decisions to alter or discontinue VTE prophylaxis should occur in consultation with the patient's oncologist.

Incidental Diagnosis and Thrombophilia Workup

Patients will occasionally be diagnosed with PE discovered incidentally during routine imaging. This is often because of a CT scan of the chest performed with iodinated contrast for other reasons, such as staging of lung cancer or routine surveillance. The prognosis and treatment for these emboli are unchanged and confer the same risk to the patient as does symptomatic embolism [36]. All cancer patients with any confirmed venous thrombosis require systemic anticoagulation if they have no contraindications. Further, risk of recurrent VTE is significant despite anticoagulant treatment [63].

The thrombophilia workup adds only unnecessary cost to the care of the cancer patient [64]. In terms of treatment choices and duration, cancer dominates as the driver of decision-making regarding type and duration of anticoagulation, regardless of the patient's other genetic predisposition to thrombosis. Current guidelines do not support the testing of patients or their families for thrombophilia in the setting of a cancer-associated thrombus, including the "Choose Wisely" points issued by the American Society of Hematology [65]. Overtesting for inherited risk factors for VTE when not indicated lead to excess costs for patients and the health care system, and reductions in unnecessary testing can ultimately reduce health care costs [66].

Catheter-Associated Thrombosis

In 18 published studies of cancer patients with peripherally inserted central catheters, 6.8% (234/3430) were reported to experience deep venous thrombosis [67]. Intraluminal occlusions are best treated by interventional radiology or other specialists with access to and experience using fibrinolytic agents (e.g., Cathflo®) for this purpose. Extraluminal venous thrombosis (a form of DVT) can be treated by either removal of the catheter or anticoagulation with the catheter in place. In contrast to other sites of deep venous thrombosis, symptomatic PE occurs <5% of patients, but more than half experience total venous obstruction, which can lead to post-thrombotic syndrome and venous scarring with permanent stenosis [68]. Moreover, peripherally inserted central catheters (PICC lines) have a higher risk of thrombosis in cancer patients than central venous catheters [67]. PICC line-related DVT is highest within the first 2 weeks of placement, and smoking and high BMI significantly contribute to the rate of DVT; meanwhile, in a similar cohort, anticoagulation did not prevent thrombotic events [69, 70]. The decision to leave or remove the thrombosed catheter should be based primarily on the degree of swelling and pain, balanced against the need for the catheter and availability of alternative source of venous access, preferably determined in conjunction with the patient's oncologist. Regardless, PICCs can be safely and successfully utilized in nonhospitalized cancer patients for both chemotherapy and parental nutrition [71].

For indwelling lines used for active chemotherapy or other ongoing treatment, anticoagulation is often the best route. If the catheter can be removed, and the patient has trivial swelling and no pain, the author's preference for subsequent treatment is a 7-day course of anticoagulation. Patients with visible swelling or pain should have 3 months of anticoagulation [29]. Prophylactic anticoagulation has shown disappointing results in both adults and children for prevention of catheter-associated thrombosis [72, 73].

Advanced Treatment

For all patients with massive PE, defined by either hypotension (systolic blood pressure <90 mmHg) or a 40 mmHg drop in systolic blood pressure (observed in the ED), clinical guidelines are aligned to recommend systemic fibrinolysis in the absence of contraindications [74, 75]. We recommend infusion of 15 mg of alteplase followed by 85 mg over 2 h. All patients should receive full-dose heparin anticoagulation (e.g., 5000 U unfractionated heparin bolus, followed by 16–18 U/kg/h infusion and PTT monitoring).

Patients with cancer have higher bleeding risk with standard anticoagulation and likely have higher bleeding risks with administration of fibrinolytic agents [76, 77]. Recent work suggests that a subpopulation of patients with DVT and PE will benefit from advanced therapies such as thrombolytics or catheter-based treatment [78, 79]. These patients fall primarily into two categories: DVT patients with large iliofemoral clot burden causing pain and leg swelling, and PE patients with right ventricular dysfunction, evidenced by an elevated troponin measurement (>99 percentile at a precision of 10% coefficient of variability), elevated brain natriuretic peptide (BNP >90 pg/mL), elevated pro-BNP (>900 pg/mL), or an echocardiogram demonstrating right ventricular hypokinesis or dilation, often defined as the right ventricular diameter larger than the left ventricular diameter (Table 24.6) [74]. Limited retrospective data exists for catheter-based

 Table 24.6
 Findings that suggest a worsened prognosis and may serve as indications for thrombolysis

Vital sign and physical	
examination abnormalities	Lab and imaging abnormalities
Systolic blood pressure <90 mmHg or a 40 mmHg drop from a known baseline blood pressure	Elevated cardiac BNP (>90 pg/mL) or pro-BNP (>900 pg/mL)
Hypoxemia (<92% at or near sea level) with respiratory distress	Elevated troponin or CK-MB
Altered mental status or delirium, usually seen as agitation, panic, and inattention	Right ventricular dilation with or without hypokinesis on echocardiography; a tricuspid annular plane systolic excursion (TAPSE) distance <16 mm; lobar or larger clot burden on CT together with RV > LV and evidence of contrast reflux

BNP brain natriuretic peptide, *CK* creatine kinase, *CT* computed tomography, *RV* right ventricle, *LV* Left ventricle

treatment of either PE or DVT in cancer patients [80]. Regarding inferior vena caval filters in patients with PE who can be anticoagulated, no evidence has shown a clinically important net benefit for their insertion, and we do not recommend their use in any patient with PE who can be anticoagulated [81].

Patients with Absolute Contraindications to Anticoagulation

The treatment options are limited. An important intervention for these patients is to insert a vena caval filter as soon as possible [82]. With limited options for the patient in the presence of a strong contraindication to anticoagulation, such as active gastrointestinal bleeding, or recent glioblastoma surgery, physicians may be forced to consider treatments based upon case reports or hypothetical reasoning. These include inhaled nitric oxide (35 ppm by face mask or 50 ppm by nasal cannula) to reduce pulmonary vascular resistance and inhaled heparin localize anticoagulation in the lung vasculature [83]. Under fluoroscopic guidance, an interventional radiologist can place an infusion catheter through the body of the thrombus lodged in a proximal pulmonary artery and infuse 0.5 mg/h of alteplase with or without adjunctive ultrasonic (e.g., Ekosonic®), hydraulic (e.g., Angiojet®), or mechanical (e.g., rotating pigtail) thrombus disruption. No clinical trials have tested this method and the risk of hemorrhage remains unknown. Methods of clot extraction, including the use of the large bore Angiovac® device or open surgical thrombectomy, require that the patient be placed on an extracorporeal perfusion circuit with systemic anticoagulation.

Discontinuation of Treatment

One of the primary challenges of treating VTE in the cancer population is choosing a stop date for the desired therapy. In noncancer patients, anticoagulation may be continued for 3-6 months [84]. Given the prothrombic nature of cancer, the exact duration of anticoagulation treatment has no welldefined criteria. Instead, knowing that the risk of recurrence of thrombosis after discontinuation of anticoagulation exists, physicians must use their best judgment as to when to discuss discontinuation. Whenever possible, patients and families should be involved in the decision-making process, as at some point continuation of anticoagulation may offer no symptomatic benefit or improvement in the qualities of life [85, 86]. Ultimately, the decision to discontinue is best made through shared decision among all involved parties, especially as performance status declines and as death approaches.

VTE as First Manifestation of Cancer

Again, VTE in the setting of known cancer will not benefit from additional thrombophilia workup; however, a newly diagnosed, unprovoked VTE may be associated with underlying cancer [87]. One prospective study found that 3.3% of newly diagnosed VTE were the result of an underlying, undiagnosed malignancy, as compared to 0% of the provoked VTE. Furthermore, the presence of residual vein thrombosis in patients with unprovoked DVT who do not develop cancer within 3 months of the diagnosis of VTE is an independent predictor of subsequent overt malignancy [88]. Several other studies have demonstrated an association between idiopathic VTE and underlying undiagnosed cancer [89, 90].

Follow-Up

Discharge of patients for outpatient follow-up must be done in conjunction with an outpatient physician. This may either be the patient's primary care physician or oncologist. Determining if the VTE is primary or secondary may provoke physicians to perform additional testing to see if an underlying malignancy is present. As stated earlier, with the development of the target specific anticoagulants, routine anticoagulation monitoring in the form of "coumadin clinics" will become less prevalent, especially in view of the fact that warfarin is not recommended to treat active cancer. Thus, the role of the physician in evaluating and caring for the patient with blood clots will shift away from simple INR management and dosage adjustment, and instead focus on duration of therapy, prognosis, and quality of life. These skills are well within the purview of primary care physicians. Stopping criteria, while imperfect, can be used in a shared decision-making model to determine individual duration of anticoagulation [57].

References

- Liebman HA. Cancer prognosis in patients with venous thromboembolism (VTE) and patients with clinical and laboratory biomarkers predictive of VTE risk. Thromb Res. 2018;164(Suppl 1):S19–22.
- Blom JW, Doggen CM, Osanto S, Rosendaal FR. Malignancies, prothrombotic mutations, and the risk of venous thrombosis. JAMA. 2005;293(6):715–22.
- Bharthuar A, Khorana AA, Hutson A, Wang JG, Key NS, Mackman N, et al. Circulating microparticle tissue factor, thromboenbolism and survival in pancreaticobiliary cancers. Thromb Res. 2013;132(2):180–4.
- Garcia Rodriguez P, Eikenboom HC, Tesselaar ME, Huisman MV, Nijkeuter M, Osanto S, et al. Plasma levels of microparticleassociated tissue factor activity in patients with clinically suspected pulmonary embolism. Thromb Res. 2010;126(4):345–9.

- Castelli R, Ferrari B, Cortelezzi A, Guariglia A. Thromboembolic complications in malignant haematological disorders. Curr Vasc Pharmacol. 2010;8(4):482–94.
- Falanga A. Mechanisms of hypercoagulation in malignancy and during chemotherapy. Haemostasis. 1998;28(Suppl 3):50–60.
- Caine GJ, Stonelake PS, Lip GY, Kehoe ST. The hypercoagulable state of malignancy: pathogenesis and current debate. Neoplasia. 2002;4(6):465–73.
- Shaw JR, Douketis J, Le Gal G, Carrier M. Periprocedural interruption of anticoagulation in patients with cancer-associated venous thromboembolism: an analysis of thrombotic and bleeding outcomes. J Thromb Haemost. 2019;17(7):1171–8.
- 9. Abdol Razak NB, Jones G, Bhandari M, Berndt MC, Metharom P. Cancer-associated thrombosis: an overview of mechanisms, risk factors, and treatment. Cancers (Basel). 2018;10(10):380.
- Chew HK, Wun T, Harvey D, Zhou H, White RH. Incidence of venous thromboembolism and its effect on survival among patients with common cancers. Arch Intern Med. 2006;166(4):458–64.
- Lee AY, Levine MN. The thrombophilic state induced by therapeutic agents in the cancer patient. Semin Thromb Hemost. 1999;25(2):137–45.
- 12. Thiruganasambandamoorthy V, Turko E, Ansell D, Vaidyanathan A, Wells GA, Stiell IG. Risk factors for serious underlying pathology in adult emergency department nontraumatic low back pain patients. J Emerg Med. 2014;47(1):1–11.
- Zhang Y, Yang Y, Chen W, Guo L, Liang L, Zhai Z. Wang C; China Venous Thromboembolism (VTE) Study Group. Prevalence and associations of VTE in patients with newly diagnosed lung cancer. Chest. 2014;146(3):650–8.
- Mahajan A, Brunson A, White R, Wun T. The epidemiology of cancer-associated venous thromboembolism: an update. Semin Thromb Hemost. 2019;45(4):321–5.
- Tang Y, Sampson B, Pack S, Shah K, Yon Um S, Wang D, et al. Ethnic differences in out-of-hospital fatal pulmonary embolism. Circulation. 2011;123(20):2219–25.
- Rodeghiero F, Castaman G, Dini E. Fibrinopeptide A changes during remission induction treatment with L-asparaginase in acute lymphoblastic leukemia: evidence for activation of blood coagulation. Thromb Res. 1990;57(1):31–8.
- Zia AN, Chitlur M. Management of thrombotic complications in acute lymphoblastic leukemia. Indian J Pediatr. 2013;80(10):853–62.
- Khorana AA, Kuderer NM, Culakova E, Lyman GH, Francis CW. Development and validation of a predictive model for chemotherapy-associated thrombosis. Blood. 2008;111(10):4902–7.
- 19. Kato A, Takano H, Ichikawa A, Koshino M, Igarashi A, Hattori K, et al. A retrospective cohort study of venous thromboembolism(VTE) in 1035 Japanese myeloma patients treated with thalidomide; lower incidence without statistically significant association between specific risk factors and development of VTE and effects of thromboprophylaxis with aspirin and warfarin. Thromb Res. 2013;131(2):140–4.
- Palumbo A, Palladino C. Venous and arterial thrombotic risks with thalidomide: evidence and practical guidance. Ther Adv Drug Saf. 2012;3(5):255–66.
- 21. Hurwitz HI, Saltz LB, Van Cutsem E, Cassidy J, Wiedemann J, Sirzen F, et al. Venous thromboembolic events with chemotherapy plus bevacizumab: a pooled analysis of patients in randomized phase II and III studies. J Clin Oncol. 2011;29(13):1757–64.
- Di Nisio M, Candeloro M, Rutjes AWS, Porreca E. Venous thromboembolism in cancer patients receiving neoadjuvant chemotherapy: a systematic review and meta-analysis. J Thromb Haemost. 2018;16(7):1336–46.
- Chee CE, Ashrani AA. Predictors of venous thromboembolism recurrence and bleeding among active cancer patients: a populationbased cohort study. Blood. 2014;123(25):3972–8.

- Hong SI, Ahn S, Lee YS, Kim WY, Lim KS, Lee JH, et al. Contrastinduced nephropathy in patients with active cancer undergoing contrast-enhanced computed tomography. Support Care Cancer. 2016;24(3):1011–7.
- Mitchell AM, Jones AE, Tumlin JA, Kline JA. Incidence of contrast-induced nephropathy after contrast-enhanced computed tomography in the outpatient setting. Clin J Am Soc Nephrol. 2010;5(1):4–9.
- Latcha S, Plodkowski AJ, Zheng J, Jaimes EA. Rate and risk factors for AKI after CT scans in a cancer cohort. Clin Nephrol. 2019;91(3):147–54.
- 27. Ay C, Dunkler D, Pirker R, Thaler J, Quehenberger P, Wagner O, et al. High D-dimer levels are associated with poor prognosis in cancer patients. Haematologica. 2012;97(8):1158–64.
- Othieno R, Abu Affan M, Okpo E. Home versus in-patient treatment for deep vein thrombosis. Cochrane Database Syst Rev. 2007;(3):CD003076.
- 29. Kearon C, Akl EA, Comerota AJ, Prandoni P, Bounameaux H, Goldhaber SZ, et al. Antithrombotic therapy for VTE disease: antithrombotic therapy and prevention of thrombosis, 9th ed: American College of Chest Physicians evidence-based clinical practice guidelines. Chest. 2012;141(2 Suppl):e419S–96S.
- 30. National Comprehensive Cancer Network. NCCN flash update: NCCN Guidelines® for lung cancer screening and cancer-associated venous thromboembolic disease and NCCN Evidence Blocks[™] for Hodgkin lymphoma. Version 2; 2019. https://www.nccn.org/about/ news/ebulletin/ebulletindetail.aspx?ebulletinid=1498. Accessed 5 May 2020.
- 31. Tzoran I, Saharov G, Brenner B, Delsart D, Román P, Visoná A, Jiménez D. Monreal M; RIETE Investigators. Silent pulmonary embolism in patients with proximal deep vein thrombosis in the lower limbs. J Thromb Haemost. 2012;10(4):564–71.
- 32. Enden T, Haig Y, Kløw N-E, Slagsvold C-E, Sandvik L, Ghanima W, et al. Long-term outcome after additional catheter-directed thrombolysis versus standard treatment for acute iliofemoral deep vein thrombosis (the CaVenT study): a randomised controlled trial. Lancet. 2012;379(9810):31–8.
- Vinson DR, Zehtabchi S, Yealy DM. Can selected patients with newly diagnosed pulmonary embolism be safely treated without hospitalization? A systematic review. Ann Emerg Med. 2012;60(5):651–62.
 e4. Erratum in: Ann Emerg Med. 2015;65(2):177.
- 34. Zondag W, Kooiman J, Klok FA, Dekkers OM, Huisman MV. Outpatient versus inpatient treatment in patients with pulmonary embolism: a meta-analysis. Eur Respir J. 2013;42(1):134–44.
- 35. Exter PL, Gomez V, Jimenez D, Trujillo-Santos J, Muriel A, Huisman MV, Monreal M. A clinical prognostic model for the identification of low-risk patients with acute symptomatic pulmonary embolism and active cancer. Chest. 2013;143(1):138–45.
- Kline JA, Miller DW. Risk stratification for acute pulmonary embolism. J Natl Compr Cancer Netw. 2011;9(7):800–10.
- 37. MacLean S, Mulla S, Akl EA, Jankowski M, Vandvik PO, Ebrahim S, et al. Patient values and preferences in decision making for antithrombotic therapy: a systematic review: antithrombotic therapy and prevention of thrombosis, 9th ed: American College of Chest Physicians evidence-based clinical practice guidelines. Chest. 2012;141(2 Suppl):e1S-e23S.
- 38. Prins MH, Lensing AW, Bauersachs R, van Bellen B, Bounameaux H, Brighton TA, et al. EINSTEIN Investigators. Oral rivaroxaban versus standard therapy for the treatment of symptomatic venous thromboembolism: a pooled analysis of the EINSTEIN-DVT and PE randomized studies. Thromb J. 2013;11(1):21.
- 39. Kline JA, Roy PM, Than MP, Hernandez J, Courtney DM, Jones AE, et al. Derivation and validation of a multivariate model to predict mortality from pulmonary embolism with cancer: the POMPE-C tool. Thromb Res. 2012;129(5):e194–9.

- 40. den Exter PL, Gomez V, Jimenez D, Trujillo-Santos J, Muriel A, Huisman MV, et al. A clinical prognostic model for the identification of low-risk patients with acute symptomatic pulmonary embolism and active cancer. Chest. 2013;143(1):138–45.
- 41. Zondag W, Hiddinga BI, Crobach MJ, Labots G, Dolsma A, Durian M, et al. Hestia criteria can discriminate high- from low-risk patients with pulmonary embolism. Eur Respir J. 2013;41(3):588–92.
- 42. Zondag W, Mos IC, Creemers-Schild D, Hoogerbrugge AD, Dekkers OM, Dolsma J, et al. Outpatient treatment in patients with acute pulmonary embolism: the Hestia Study. J Thromb Haemost. 2011;9(8):1500–7.
- 43. Beam DM, Kahler ZP, Kline JA. Immediate discharge and home treatment with rivaroxaban of low-risk venous thromboembolism diagnosed in two U.S. emergency departments: a one-year preplanned analysis. Acad Emerg Med. 2015;22(7):788–95.
- 44. Ahn S, Cooksley T, Banala S, Buffardi L, Rice TW. Validation of the EPIPHANY index for predicting risk of serious complications in cancer patients with incidental pulmonary embolism. Support Care Cancer. 2018;26(10):3601–7.
- 45. Lee AYY, Levine MN, Baker RI, Bowden C, Kakkar AK, Prins M, et al. Low-molecular-weight heparin versus a coumarin for the prevention of recurrent venous thromboembolism in patients with cancer. N Engl J Med. 2003;349(2):146–53.
- 46. Noble S, Matzdorff A, Maraveyas A, Holm MV, Pisa G. Assessing patients' anticoagulation preferences for the treatment of cancerassociated thrombosis using conjoint methodology. Haematologica. 2015;100(11):1486–92.
- Bossaer JB, Covert KL. Direct oral anticoagulants in patients with cancer. Am J Health Syst Pharm. 2019;76(14):1019–27.
- 48. Kahale LA, Hakoum MB, Tsolakian IG, Matar CF, Terrenato I, Sperati F, et al. Anticoagulation for the long-term treatment of venous thromboembolism in people with cancer. Cochrane Database Syst Rev. 2018;6:CD006650.
- 49. Young AM, Marshall A, Thirlwall J, Chapman O, Lokare A, Hill C, et al. Comparison of an oral factor Xa inhibitor with low molecular weight heparin in patients with cancer with venous thromboembolism: results of a randomized trial (SELECT-D). J Clin Oncol. 2018;36(20):2017–23.
- 50. Prins MH, Lensing AW, Brighton TA, Lyons RM, Rehm J, Trajanovic M, et al. Oral rivaroxaban versus enoxaparin with vitamin K antagonist for the treatment of symptomatic venous thromboembolism in patients with cancer (EINSTEIN-DVT and EINSTEIN-PE): a pooled subgroup analysis of two randomised controlled trials. Lancet Haematol. 2014;1(1):e37–46.
- Vedovati MC, Giustozzi M, Becattini C. Venous thromboembolism and cancer: current and future role of direct-acting oral anticoagulants. Thromb Res. 2019;177:33–41. https://doi.org/10.1016/j. thromres.2019.02.031.
- 52. Cano SJ, Lamping DL, Bamber L, Smith S. The Anti-Clot Treatment Scale (ACTS) in clinical trials: cross-cultural validation in venous thromboembolism patients. Health Qual Life Outcomes. 2012;10:120.
- 53. Prins MH, Bamber L, Cano SJ, Wang MY, Erkens P, Bauersachs R, et al. Patient-reported treatment satisfaction with oral rivaroxaban versus standard therapy in the treatment of pulmonary embolism; results from the EINSTEIN PE trial. Thromb Res. 2015;135(2):281–8.
- 54. Fumagalli S, Cardini F, Roberts AT, Boni S, Gabbai D, Calvani S, et al. Psychological effects of treatment with new oral anticoagulants in elderly patients with atrial fibrillation: a preliminary report. Aging Clin Exp Res. 2015;27(1):99–102.
- 55. Afzal SK, Hasan SS, Babar ZU. A systematic review of patientreported outcomes associated with the use of direct-acting oral anticoagulants. Br J Clin Pharmacol. 2019;85(12):2652–67.
- 56. Debourdeau P, Arvers P, Hij A, Bennani H, Desauw C, Falvo N, et al. Factors influencing patient's perception of long-term treat-

ment with low-molecular-weight heparins for cancer-associated thrombosis: an updated analysis of TROPIQUE, a prospective observational study. Support Care Cancer. 2020;28(1):287–93.

- Ageno W, Dentali F, Donadini MP, Squizzato A. Optimal treatment duration of venous thrombosis. J Thromb Haemost. 2013;11(Suppl 1):151–60.
- Khorana AA, Soff GA, Kakkar AK, Vadhan-Raj S, Riess H, Wun T, et al. Rivaroxaban for thromboprophylaxis in high-risk ambulatory patients with cancer. N Engl J Med. 2019;380(8):720–8.
- Carrier M, Abou-Nassar K, Mallick R, Tagalakis V, Shivakumar S, Schattner A, et al. Apixaban to prevent venous thromboembolism in patients with cancer. N Engl J Med. 2019;380(8):711–9.
- 60. Agnelli G, Gussoni G, Bianchini C, Verso M, Mandala M, Cavanna L, et al. Nadroparin for the prevention of thromboembolic events in ambulatory patients with metastatic or locally advanced solid cancer receiving chemotherapy: a randomised, placebo-controlled, double-blind study. Lancet Oncol. 2009;10(10):943–9.
- Agnelli G, George DJ, Kakkar AK, Fisher W, Lassen MR, Mismetti P, et al. Semuloparin for thromboprophylaxis in patients receiving chemotherapy for cancer. N Engl J Med. 2012;366(7):601–9.
- 62. Key NS, Khorana AA, Kuderer NM, Bohlke K, Lee AYY, Arcelus JI, et al. Venous thromboembolism prophylaxis and treatment in patients with cancer: ASCO clinical practice guideline update. J Clin Oncol. 2019:JCO1901461.
- 63. Kraaijpoel N, Bleker SM, Meyer G, Mahe I, Munoz A, Bertoletti L, et al. Treatment and long-term clinical outcomes of incidental pulmonary embolism in patients with cancer: an international prospective cohort study. J Clin Oncol. 2019;37(20):1713–20.
- 64. De Stefano V, Rossi E. Testing for inherited thrombophilia and consequences for antithrombotic prophylaxis in patients with venous thromboembolism and their relatives. A review of the guidelines from Scientific Societies and Working Groups. Thromb Haemost. 2013;110(4);697–705.
- 65. American Society of Hematology. Choosing wisely. Ten things physicians and patients should question. Released 4 Dec 2013 (1–5) and 3 Dec 2014 (6–10); last reviewed 2019. Philadelphia: ABIM Foundation; 2014. choosingwisely.org; https://www.choosingwisely.org/societies/american-society-of-hematology/. Accessed 5 May 2020.
- 66. Virparia R, Brunetti L, Vigdor S, Adams CD. Appropriateness of thrombophilia testing in patients in the acute care setting and an evaluation of the associated costs. J Thromb Thrombolysis. 2020;49(1):108–12.
- 67. Chopra V, Anand S, Hickner A, Buist M, Rogers MA, Saint S, et al. Risk of venous thromboembolism associated with peripherally inserted central catheters: a systematic review and meta-analysis. Lancet. 2013;382(9889):311–25.
- Farrell JJ, Sutter C, Tavri S, Patel I. Incidence and interventions for post-thrombotic syndrome. Cardiovasc Diagn Ther. 2016;6(6):623–31.
- 69. Al-Asadi O, Almusarhed M, Eldeeb H. Predictive risk factors of venous thromboembolism (VTE) associated with peripherally inserted central catheters (PICC) in ambulant solid cancer patients: retrospective single Centre cohort study. Thromb J. 2019;17:2.
- 70. Jones D, Wismayer K, Bozas G, Palmer J, Elliott M, Maraveyas A. The risk of venous thromboembolism associated with peripherally inserted central catheters in ambulant cancer patients. Thromb J. 2017;15:25.
- Cotogni P, Barbero C, Garrino C, Degiorgis C, Mussa B, De Francesco A, et al. Peripherally inserted central catheters in nonhospitalized cancer patients: 5-year results of a prospective study. Support Care Cancer. 2015;23(2):403–9.
- Brandao LR, Shah N, Shah PS. Low molecular weight heparin for prevention of central venous catheterization-related thrombosis in children. Cochrane Database Syst Rev. 2014;3:CD005982.

- 73. Akl EA, Labedi N, Terrenato I, Barba M, Sperati F, Sempos EV, et al. Low molecular weight heparin versus unfractionated heparin for perioperative thromboprophylaxis in patients with cancer. Cochrane Database Syst Rev. 2011;(11):CD009447.
- 74. Jaff MR, McMurtry MS, Archer SL, Cushman M, Goldenberg N, Goldhaber SZ, et al. Management of massive and submassive pulmonary vein thrombosis, and chronic thromboembolic pulmonary hypertension: statement from the American Heart Association. Circulation. 2011;1:1788–830.
- 75. Holbrook A, Schulman S, Witt DM, Vandvik PO, Fish J, Kovacs MJ, et al. Evidence-based management of anticoagulant therapy: antithrombotic therapy and prevention of thrombosis, 9th ed: American College of Chest Physicians evidence-based clinical practice guidelines. Chest. 2012;141(2 Suppl):e152S–84S.
- 76. Trujillo-Santos J, Nieto JA, Tiberio G, Piccioli A, Di Micco P, Prandoni P, et al. Predicting recurrences or major bleeding in cancer patients with venous thromboembolism. Findings from the RIETE Registry. Thromb Haemost. 2008;100(3):435–9.
- Gross CP, Galusha DH, Krumholz HM. The impact of venous thromboembolism on risk of death or hemorrhage in older cancer patients. J Gen Intern Med. 2007;22(3):321–6.
- Enden T, Haig Y, Klow NE, Slagsvold CE, Sandvik L, Ghanima W, et al. Long-term outcome after additional catheter-directed thrombolysis versus standard treatment for acute iliofemoral deep vein thrombosis (the CaVenT study): a randomised controlled trial. Lancet. 2012;379(9810):31–8.
- Kucher N, Boekstegers P, Muller OJ, Kupatt C, Beyer-Westendorf J, Heitzer T, et al. Randomized, controlled trial of ultrasoundassisted catheter-directed thrombolysis for acute intermediate-risk pulmonary embolism. Circulation. 2014;129(4):479–86.
- Maleux G, Marchal P, Palmers M, Heye S, Verhamme P, Vaninbroukx J, et al. Catheter-directed thrombolytic therapy for thoracic deep vein thrombosis is safe and effective in selected patients with and without cancer. Eur Radiol. 2010;20(9):2293–300.
- Coombs C, Kuk D, Devlin S, Siegelbaum RH, Durack JC, Parameswaran R, et al. Outcomes after inferior vena cava filter placement in cancer patients diagnosed with pulmonary embolism: risk for recurrent venous thromboembolism. J Thromb Thrombolysis. 2017;44(4):489–93.
- Harvey JJ, Hopkins J, McCafferty IJ, Jones RG. Inferior vena cava filters: what radiologists need to know. Clin Radiol. 2013;68(7):721–32.
- Kline JA, Hernandez J, Garrett JS, Jones AE. Pilot study of a protocol to administer inhaled nitric oxide to treat severe acute submassive pulmonary embolism. Emerg Med J. 2014;31(6):459–62.
- 84. British Thoracic Society Standards of Care Committee Pulmonary Embolism Guideline Development Group. British Thoracic Society guidelines for the management of suspected acute pulmonary embolism. Thorax. 2003;58(6):470–83.
- Noble S. Venous thromboembolism and palliative care. Clin Med (Lond). 2019;19(4):315–8.
- Noble S. The challenges of managing cancer related venous thromboembolism in the palliative care setting. Postgrad Med J. 2007;83(985):671–4.
- Prandoni P, Lensing AW, Buller HR, Cogo A, Prins MH, Cattelan AM, et al. Deep-vein thrombosis and the incidence of subsequent symptomatic cancer. N Engl J Med. 1992;327(16):1128–33.
- Prandoni P, Lensing AW, Prins MH, Villalta S, Negri S, Piccioli A, et al. Venous thrombotic burden and the risk of subsequent overt cancer. Thromb Haemost. 2015;114(4):864–5.
- Monreal M, Lafoz E, Casals A, Inaraja L, Montserrat E, Callejas JM, et al. Occult cancer in patients with deep venous thrombosis. A systematic approach. Cancer. 1991;67(2):541–5.
- Monreal M, Prandoni P. Venous thromboembolism as first manifestation of cancer. Semin Thromb Hemost. 1999;25(2):131–6.

Gastroenterology

25

Jessica X. Yu, Jennifer L. Maranki, Gene Bakis, and Brintha K. Enestvedt

Case Study

A 65-year-old male with 2-year history of diabetes presents to his local emergency department with a history of progressive anorexia, early satiety, and 2 days of jaundice. Work up revealed a bilirubin of 5.3 (high) and a CAT scan of the abdomen and pelvis showed a markedly dilated stomach that was fluid and food filled. A 3-cm solid enhancing mass was seen in the head of the pancreas resulting in upstream biliary obstruction, duodenal invasion, and pancreatic duct dilation. A nasogastric tube was placed in the emergency room to suction with prompt evacuation of 2.5 L of dark brown murky fluid and relief of many of his symptoms. Gastroenterology was consulted.

The following day, upper endoscopy revealed a highgrade ulcerated duodenal stricture at the apex of the duodenal bulb. The endoscope was unable to be advanced to the second portion of the duodenum due to the stricture. Endoscopic ultrasound was performed, and fine needle aspiration of the pancreatic mass was positive for malignancy per the rapid on site cytopathologist. A duodenal stent was placed traversing the stricture. After the enteral stent was allowed to expand over the next 2 days to facilitate endoscopic access to the major papilla, ERCP was performed with placement of a biliary stent across the malignant biliary stricture. Staging CT scans of the chest, abdomen, and pelvis were performed and demonstrated locally advanced pancreatic cancer. A multidisciplinary pancreatic oncologic team met and offered the patient neoadjuvant chemotherapy and

Division of Gastroenterology and Hepatology, Department of Medicine, Oregon Health and Science University, Portland, OR, USA e-mail: yujess@ohsu.edu

J. L. Maranki

staging CT scans to follow to assess for surgical resectability.

This case illustrates the use of advanced endoscopic techniques to diagnose and palliate symptoms of duodenal and biliary obstruction due to a new tumor diagnosis.

Introduction

In the past, gastroenterology's involvement in oncologic care centered on endoscopic tissue acquisition for malignant diagnoses. With technological advances in gastroenterology (GI), the field of interventional endoscopy has increasingly become married to cancer care from staging to surveillance to management of tumor or treatment-related emergencies. Oncologic emergencies in gastroenterology, much like that of other body systems, are related to the presence of cancer (the tumor itself or metastatic growth) or occur as a consequence of the treatments undertaken for cancer. In the GI tract, these include both structural derangements such as luminal or biliary obstruction as well as metabolic emergencies such as hepatic failure. In general, GI-related emergencies represent a minority of all oncologic emergencies, which are more often metabolic or hematologic. Many GI-related oncologic events require timely evaluation; however, for the most part, they are not imminently life threatening. These events will be the focus of this chapter. Some types of GI oncologic emergencies are insidious and develop gradually over time until a clinical threshold is achieved (e.g., gastric outlet obstruction, jaundice), whereas others may manifest over hours (e.g., cholangitis related to biliary obstruction).

In some patients, the devastating emergency is the first presentation of cancer itself. Given the increase in outpatient oncologic care in modern-day practice, early signs and symptoms of an evolving gastrointestinal emergency or urgency may be overlooked by patients and families. As with any emergency, early recognition and diagnosis are paramount to effective management. Early consultation with a gastroenterologist to help in diagnosis and management may

J. Yu $(\boxtimes) \cdot G$. Bakis \cdot B. K. Enestvedt

Division of Gastroenterology and Hepatology, Department of Medicine, Penn State Hershey Medical Center, Penn State College of Medicine, Hershey, PA, USA

K. H. Todd et al. (eds.), Oncologic Emergency Medicine, https://doi.org/10.1007/978-3-030-67123-5_25

J. Yu et al.

 Table 25.1 The most common oncologic emergencies in gastroenterology

Gastrointestinal hemorrhage
Luminal obstruction
Acute pancreatitis
Biliary obstruction/cholangitis
Hepatic decompensation
Dysfunction of enteral feeding devices

impact patient outcome, and thus, multidisciplinary care for complex oncologic patients serves the patient best. This chapter will examine the most common cancer-related gastrointestinal emergencies, discuss their diagnosis, and review their treatment (Table 25.1). Specifically, we will address gastrointestinal hemorrhage, enteral and biliary obstruction, acute pancreatitis, hepatic decompensation, and urgent issues related to enteral feeding.

Gastrointestinal Hemorrhage

Bleeding Related to Tumors

Neoplasia is considered an infrequent cause of upper gastrointestinal bleeding (UGIB), accounting for approximately 5% of all UGIB cases [1–4]. While oncologic patients are at risk for common causes of gastrointestinal bleeding (peptic ulcer disease, gastroduodenal erosions, esophagitis, etc.), those patients that have solid tumors of the gastrointestinal tract, or metastatic lesions that involve the GI mucosa, pose a bleeding threat from the esophagus to the anus. This risk is further compounded by the presence of neutropenia and thrombocytopenia, as well as the use of anticoagulation in the setting of tumor-related thromboembolic disease. These factors, as well as the innate characteristics of the tumor, can make endoscopic treatment of bleeding a significant management challenge. Additionally, cancer-related bleeding overall is a poor prognostic sign.

A large prospective study of over 3200 patients admitted with acute nonvariceal upper GI bleeding demonstrated that those patients with neoplasia had 2.5 times (95% CI, 1.32– 4.46; P < 0.0001) the risk of death when compared to those who bled from benign conditions [1]. Of those with cancer (n = 153), the mortality rate was significantly higher in those with esophageal compared to cancer of the gastric cardia or gastric body (33% vs. 23.5% and 7.2%, respectively) [1].

Lesions in the esophagus (squamous cell carcinoma and adenocarcinoma), stomach (adenocarcinoma, gastrointestinal stromal tumors [GIST], lymphoma, carcinoid tumors, etc.), and proximal small bowel (adenocarcinoma, lymphoma, GIST, carcinoid tumors, etc.) are typically reached with a standard upper endoscope, while tumors in the jejunum and ileum often require the use of device-assisted enteroscopy, mainly in the form of single- and double-balloon enteroscopy or spiral enteroscopy. Colonic (adenocarcinoma, neuroendocrine [including carcinoid] tumors, etc.) and terminal ileal (adenocarcinoma, lymphoma) lesions are accessed via standard colonoscopy. Irrespective of the direction of endoscopic approach or the specific endoscope utilized, catheter-based devices (clips, injection needles, cautery devices) used to treat bleeding are advanced down the working channel of the endoscope and directed at the target lesion.

A variety of endoscopic techniques are available for the treatment of GI bleeding, and these modalities may also be applied to bleeding tumors although overall success rates of hemostasis are inferior compared to the endoscopic treatment of benign bleeding pathology. The main reason for the limited success of endoscopic therapy for tumor-related bleeding is that bleeding in such cases is not generally from a single exposed vessel (for which the majority of current endoscopic treatments are designed). Malignant bleeding tends to be diffuse mucosal oozing from numerous small microvessels, and therefore focal targeting of therapy is less effective due to the generally larger surface area involved.

The literature on the efficacy of endoscopic therapy for bleeding directly due to the primary or metastatic malignancy is sparse and is mainly based on case reports and series [5, 6]. Thermal therapies, including heater probe, bipolar electrocautery, and argon plasma coagulation (APC), are perhaps the most widely used modalities for tumor bleeding [7]. There have been a few case reports in the literature, but no trials comparing the efficacy of bipolar electrocautery and heater probe. A 1996 study by Savides and colleagues reported on the use of heater probe or bipolar electrocautery with or without epinephrine injection, in seven patients with focally oozing tumors. Initial hemostasis was achieved in all, but the 30-day rebleeding rate was 29%, similar to those not treated endoscopically [8]. Several studies have evaluated the use of neodymium-yttrium aluminum garnet (Nd:YAG) laser for tumor palliation in the GI tract [9-13]. Immediate hemostasis rates of 94% were achieved with the Nd: YAG laser for the emergent treatment of 18 patients with massive bleeding from esophageal and gastric cancer; however, rebleeding occurred in three of 17 patients with initial success, making laser therapy less suitable for definitive therapy [10].

Argon plasma coagulation (APC) is widely accepted as an effective modality for hemostasis as well as tissue fulguration and is readily available in most endoscopy suites and hospitals. It has generally replaced laser therapy due to its ease of use, low cost, and portability. The specific use of APC for palliation of upper GI tumor has not been well studied, with very few small case series [14–16] related to its use in esophagogastric or rectosigmoid cancers. Much of the literature related to APC in the setting of malignancy focuses on its use as a curative therapy for early-stage cancers (i.e., treatment of high-grade dysplasia in the setting of Barrett's esophagus, which has now been largely replaced by radiofrequency ablation or cryotherapy) or for decreasing tumor bulk to maintain luminal patency for palliation of dysphagia in esophageal cancer, which is discussed separately.

Spray cryotherapy is a newer endoscopic therapy that has been mainly used for the ablation of dysplasia associated with Barrett's esophagus (BE). Cryotherapy is a noncontact, targeted application of a cryogen (liquefied nitrogen or carbon dioxide) to destroy tissue through repeated freeze/thaw cycles [17]. Cryotherapy has been shown to completely eradicate high-grade dysplasia associated with BE in 97% of treated subjects [18]. This technology has also been applied to bleeding tumors with a single case report demonstrating hemostasis achieved in locally unresectable hemorrhagic esophageal cancer [19]. A major advantage of cryotherapy over other catheter-based endoscopic therapies for the treatment of bleeding from cancer is the surface area of coverage with cryotherapy treatment (several square centimeters) compared to the focal treatment effect (several square millimeters) of other probe-based endoscopic therapies (such as heater probe, bipolar electrocoagulation, hemostatic clips). It is expected that more literature will be published on its use in malignancy given its ease of use and wide surface area coverage.

A novel modality for the treatment of gastrointestinal bleeding is the use of hemostatic powders. These inorganic powders are designed to control active bleeding by two mechanisms: (1) adhering to the bleeding site and forming a mechanical barrier when in contact with blood, tissue, and the extracellular matrix; and (2) increasing local concentration of clotting factors while enhancing clot formation [20, 21]. Many of these products have been utilized in the military for temporizing battlefield-related injuries and show great hemostatic promise. There are currently three hemostatic powders available for endoscopic use, though only the TC-325 (Hemospray[™], Cook Medical Inc., Winston-Salem, NC, USA) has FDA approval. Of all GI endoscopic therapies, hemostatic powders have thus far demonstrated the greatest potential for sustained hemostasis in malignancy. Several small case series demonstrate successful immediate and sustained hemostasis with hemostatic powders [22-25]. A recent pilot randomized control study also noted 90% immediate hemostasis rates and significantly lower rebleed rates compared to controls [26].

Mechanical methods of endoscopic hemostasis are in the form of hemoclips and are currently in use, with overall relatively poor results due to their focal targeted area of therapy. Once again, most tumor-related bleeding is diffuse mucosal hemorrhage rather than from a single exposed blood vessel. Hemostatic clips have most commonly been used for prevention of bleeding following endoscopic mucosal resection, but can be applied to neoplastic lesions. Cheng et al., reported two cases of hemostasis achieved by hemoclip placement on bleeding gastrointestinal stromal tumors [27]. Oftentimes, the bleeding tumor has several areas of active oozing, and mechanical disruption of the friable tumor with the use of a hemoclip may potentially worsen bleeding. Hemoclips are generally avoided in bleeding from tumors unless there is a focal targeted area of hemorrhage.

If endoscopic therapy fails to achieve hemostasis, angiographic evaluation by interventional radiology (IR) is generally the next step. While data on angiographic treatment of bleeding tumors is limited, angiography should be attempted as second-line therapy [28]. Access to the bleeding vessel may be achieved by selective catheterization in the distribution of the culprit vessel as evidenced by active contrast extravasation. Superselective embolization using microcoils or Gelfoam is highly successful in achieving hemostasis when the anatomy is favorable. However, given the nature of cancer-related bleeding, that is, diffuse mucosal hemorrhage from multiple microvessels, embolization of the feeding blood vessel may place the patient at risk for ischemia from collateral damage.

In those instances when both endoscopic and interventional radiology interventions are not successful in achieving hemostasis or not possible, targeted radiation therapy (RT) should be considered. Overall, RT plays a crucial role in the treatment of gastrointestinal malignancies and in the management of hemostasis. The effects of RT bleeding can be realized within a matter of days and a few fractions of treatment. In general, even though bleeding ceases, further treatment is undertaken to sustain a more durable response. RT causes damage to the intima of the blood vessels that supply the tumor, leading to capillary necrosis and thrombosis, and consequently hemostasis. This effect on tumor blood vessels as well as destruction of the tumor is the main mechanism by which RT is effective for hemostasis. Several retrospective studies on the role of RT on bleeding gastric cancers have demonstrated that palliative short-course RT is effective in hemostasis in 50–95% of cases [29].

In general, palliative surgical resection is the last option for definitive care of GI bleeding related to cancers and is associated with a poor prognosis [30, 31]. A comparison of elective and emergency presentation of gastric cancer in 291 patients reported that overall 2-year survival was less in those that presented with emergency complications requiring operative intervention (25% in emergency group versus 67% who presented electively) [31]. Surgical resection is only appropriate for surgically fit patients, and prior literature has suggested a mortality rate of 10% and morbidity rate of up to 30% in cancer patients [32–34]. Surgery indeed is effective in hemostasis; however, improvement in survival is extremely low at 6%, and the impact on the quality of life after palliative resection is not clear [33]. To date, there have been no prospective trials comparing endoscopic therapy, surgery or RT in the treatment of bleeding related to luminal tumor. Therefore, there is no clear answer as to which is the most effective treatment option. In general, the management approach is to enlist the least invasive method of hemostasis first, which may include the following sequence: endoscopy, then interventional radiology, then radiation therapy, then ultimately, surgery if necessary.

Bleeding as a Consequence of Treatment

Mallory-Weiss Tear. In patients receiving chemotherapy, nausea, vomiting, and retching are common. These symptoms place patients at risk for an upper GI bleeding related to a Mallory-Weiss tear, a mucosal injury at the gastroesophageal junction area that exposes a bleeding arterial vessel. Patients often present with hematemesis that may be clinically significant. This type of injury can produce significant bleeding especially in the setting of chemotherapy-induced thrombocytopenia. This condition can often be managed with endoscopic therapy alone with the use of electrocautery, epinephrine injection (for vasoconstriction and tamponade), or mechanical clips for hemostasis via upper endoscopy. If endoscopic therapy is unsuccessful in establishing hemostasis, IR-angiography is the next step for arterial embolization.

Radiation Proctitis. Radiation injury to the rectum and sigmoid may result from treatment of cancers of the prostate, cervix, rectum, anus, urinary bladder, and testes, occurring in up to 20% of these treated cancer patients. Endoscopic findings are mucosal ulceration, edema, erythema, bleeding, and telangiectasia from neovascularization (Fig. 25.1). It presents as persistent or intermittent hematochezia that may be associated with tenesmus or abdominal cramping and can be a significant source of GI blood loss. Acute radiation proctitis occurs either during or within 6 weeks of RT and is usually self-limited, while chronic radiation proctitis often occurs several months to years following RT [35]. Risk factors include the total surface area of exposure, method of delivery and cumulative radiation dose, with doses above 70 Gy resulting in long-term injury to the rectal mucosa [36]. Additional factors associated with the development of RT proctitis include prior GI or genitourinary surgery, endometriosis, diabetes, hypertension, collagen vascular disorders, and inflammatory bowel disease, all of which contribute to vascular compromise in the area [37]. Most cases of radiation proctitis are self-limiting and respond to medical therapy. In general, rates of radiation proctitis may be decreasing as RT techniques are improving to allow more targeted delivAV 1/20 AV 1/20 AV

Fig. 25.1 Radiation proctitis. Endoscopic appearance of radiation proctitis, characterized by multiple telangiectasias in the rectum as seen on retroflexion in sigmoidoscopy. (Courtesy of Jeffrey L. Tokar, MD, Fox Chase Cancer Center, Philadelphia, Pennsylvania)

ery of higher doses of radiation. While supportive care is typically offered for mild cases, patients who develop persistent outlet-type bleeding may benefit from targeted therapy. There is a paucity of well-designed trials comparing medical and endoscopic therapy, making it impossible to identify the most effective approach for the management of chronic radiation proctitis.

Treatment for radiation proctitis is medical therapy or endoscopic therapy. Medical therapy options include antiinflammatory agents such as 5-aminosalicylic acid containing medications; antioxidants such as vitamin A, E, and C; sucralfate and short chain fatty acid enemas; and metronidazole and formalin therapy [38]. Kochhar and colleagues reported the use of sucralfate enemas in 26 patients, with durable remission of symptoms in a majority of patients with moderate-to-severe bleeding [39]. Several other reports have also demonstrated efficacy in improving symptoms of proctitis or proctosigmoiditis [39–41]. Given its overall low cost, minimal side-effect profile, and ease of administration, sucralfate topical therapy is a reasonable first step.

Endoscopic therapy with argon plasma coagulation (APC) is currently the first-line endoscopic modality for treatment of bleeding associated with radiation proctitis. It is easy to use, effective, widely available, safe, and relatively inexpensive. The goal of endoscopic therapy to is obliterate the telangiectasias. APC provides a predictable, noncontact, uniform, limited depth of coagulation (0.5–3 mm) to the target tissue, resulting in lower risks of perforation, stricture, and fistula formation [42, 43]. Studies have also demonstrated sustained remission of bleeding in patients with severe radiation proctitis (90% in a mean follow-up of 18 months) [44]. Typically, more than one session of APC



Fig. 25.2 Palliation of duodenal (**a**) and biliary obstruction (**a**, **b**) from pancreatic cancer. (Courtesy of Vinay Chandrasekhara, MD, University of Pennsylvania, Philadelphia, PA, and Kaveh Sharzehi, MD, Oregon Health & Science University). Two example fluoroscopic images of dual duodenal and biliary stenting (**a**, **b**)

therapy is required, with durable hemostasis achieved after three sessions [38, 45].

Other endoscopic treatments include radio-frequency ablation and cryotherapy [46]. Both these therapies have the advantage of allowing a broader field of treatment than the focal therapy of APC. Initial studies have demonstrated high rates of hemostasis (>90%) as well as minimal to no side effects (up to 19 months of follow-up) with both RFA and cryotherapy [47–49], though all studies are small. RFA has the benefit of inducing neosquamous epithelialization that may prevent recurrence of symptoms, but this has yet to be studied formally [43]. Hyperbaric oxygen therapy (HBOT) is currently utilized for the treatment of radiation proctitis in those patients who have failed more conventional medical or endoscopic techniques. It involves inhalation of 100% oxygen, delivered daily for weeks in a full body chamber with increased atmospheric pressure. It has been shown to be 89% effective with minimal morbidity with overall good compliance with treatment [50]. However, this therapy is not widely available and is often associated with wound care centers.

Luminal Obstruction

Patients with GI malignancies are at risk for obstruction of the GI tract, which is the most common surgical emergency encountered in the cancer patient [51]. Obstruction is characterized by poor oral intake due to nausea, vomiting, abdominal pain, and paucity of stool or flatus passage. Abdominal X-rays or CT scans may reveal air-fluid levels and a transition point indicating the site of obstruction. Initial management is conservative, with bowel rest, intravenous fluids, antiemetics, electrolyte repletion, and nasogastric (NG) tube placement for decompression. In general, the majority of noncancer-related bowel obstructions related to adhesions resolve in this conservative approach.

Malignant obstructions of the esophagus, stomach, duodenum, and colon can be alleviated by placing a selfexpanding metal stent (SEMS) via endoscopy. Stents are placed with endoscopic and fluoroscopic guidance and are generally effective at maintaining patency in this setting (Fig. 25.2). SEMS has been shown to be a safe and effective means of maintaining esophageal patency in the setting of malignancy and the procedure has a very high technical success rate. Stent migration is the most common complication, with widely varying rates. While patients should be made aware of this potential in the informed consent process, the advent of stents with antimigration struts and endoscopic suturing devices, this risk has been reduced significantly [52, 53]. Post-procedure pain can be quite significant and is common, but often resolves after 48-72 h, with oral analgesia as support. Other considerations are the management of the induced acid reflux of the gastroesophageal junction that is bridged with the stent. As such, patients with esophageal stents are generally provided with a twice daily proton pump inhibitor (PPI) as well as instructed to sleep at a 45° angle indefinitely. The decision to place a SEMS for esophageal cancer is institution dependent and best made in a multidisciplinary setting. In some cases, surgeons prefer not to have an esophageal stent in situ if the patient is a candidate for an esophagectomy and instead prefer to provide nutrition via a surgically placed jejunostomy tube (at which time, surgical ischemic conditioning of the future gastric conduit can be performed). Esophageal stents, in this instance, are reserved for palliative cases only. Alternatively, some practices routinely place SEMS for obstructing esophageal cancer to aid in nutritional improvement prior to neoadjuvant therapy and surgery.

Gastroduodenal stenting for malignant obstruction of gastric, duodenal, or pancreatobiliary cancers has technical success rates of 90–95% according to multicenter retrospective studies [54], resulting in the ability to tolerate an enteral diet. Reintervention rates are low (only 5%), providing evidence that enteral stents achieve excellent palliation. Overall prognosis should be considered prior to palliative luminal stenting. If patient life expectancy is anticipated to exceed 6 months, a more durable approach such as a surgical bypass (gastrojejunostomy) should be employed. While generally well tolerated, the main complications of gastroduodenal stenting include stent migration, perforation, and stent obstruction by tumor or food. These obstructions can typically be resolved endoscopically by placing a stent within the originally placed stent [55–57]. GI has witnessed multiple endoscopic advances with the use of endoscopic ultrasound and one of them has been the endoscopic gastrojejunostomy in which a lumen apposing metal stent is used to create a stable fistula between the stomach and jejunum. This novel technique is being refined currently but shows promise to potentially avoid surgeries in patients with advanced cancers [58–60].

Simultaneous biliary and gastroduodenal stenting is performed relatively commonly and has been shown to be a safe and effective means of palliation [61, 62] (Fig. 25.3). An additional consideration is that luminal stenting may not be effective for infiltrative-type gastric cancers in which motility is significantly compromised due to tumor (i.e., linitis plastica), rather than a true obstruction.

Finally, colonic stents have been used in the setting of colonic obstruction for palliation and also for colonic decom-

pression prior to surgical intervention (so-called bridge to surgery). Once again, similar to other luminal stenting, technical (>92%) and clinical success rates (>89%) are high, and adverse rates are generally low, although some studies raise concerns about perforation [63–65]. There are multiple studies evaluating the role of colonic stents with respect to bridge to surgery or allowing for a one-step surgery (i.e., primary anastomosis versus an ostomy with subsequent takedown and anastomosis). Overall, results from the available literature are highly variable largely due to multiple small studies, lack of homogeneity in patient populations, as well as definitions of success and treatments. The most recent metaanalysis of eight RCTs and 497 patients in the surgical literature shows that stent as a bridge to surgery over emergency surgery for left-sided malignant colonic obstruction was associated with lower 60-day morbidity and lower rates of stoma formation; however, 30-day mortality rates are no different than surgery [66].

For surgically unresectable disease, with a life expectancy <6 months, colonic stenting is the treatment of choice [67, 68]. In potentially curable disease, colonic stents should only be considered when surgical intervention is anticipated to shortly follow stent placement. A multidisciplinary discussion between the performing gastroenterologist, the colorectal surgeon, and oncologist is warranted to optimize patient outcomes, given the lack of clear data in this area [69].



Fig. 25.3 Palliation of duodenal (**a**) and biliary obstruction (**a**, **b**) from pancreatic cancer. (Courtesy of Vinay Chandrasekhara, MD, University of Pennsylvania, Philadelphia, PA, and Kaveh Sharzehi, MD, Oregon

Health & Science University). Two example fluoroscopic images of dual duodenal and biliary stenting $({\bf a}, {\bf b})$

Acute Pancreatitis

In patients with cancer, acute pancreatitis may occur as a result of pancreatic duct obstruction from tumor or as a result of chemotherapeutic agents. The diagnosis of acute pancreatitis is established by two of the following: clinical symptoms of abdominal pain; elevation of serum pancreatic enzymes greater than three times the upper limit of normal; and imaging studies documenting active pancreatic inflammation (CT or MRI with or without intravenous contrast). Mechanical obstruction of the main pancreatic duct or its branches is one of the etiologies of acute pancreatitis. While the most common reason for such an obstruction is gallstone disease, pancreatic neoplasms are recognized as an important, albeit more rare, cause of acute pancreatitis [70]. The risk of pancreatic adenocarcinoma after a single episode of acute pancreatitis is increased, with diagnosis that is often delayed for up to 2 years [71]. Age greater than 50 years, a history of smoking, weight loss of 10 lb or greater, serum bilirubin of 2 mg/dL, or alkaline phosphatase level greater than 165 U/mL, as well as radiologic findings of distal pancreatic atrophy or mass had statistically significant associawith the subsequent diagnosis of pancreatic tion adenocarcinoma on follow-up [72]. Pancreatic neuroendocrine tumors are an uncommon cause of acute pancreatitis, but also should be included in the differential diagnosis of acute pancreatitis, especially in patients older than 40 years in whom the etiology of acute pancreatitis is not clear [73].

Acute pancreatitis can sometimes be the first presentation of primary pancreatic or ampullary neoplasms or metastatic disease to the pancreas [74, 75], the latter of which has been described in patients with cancers of the lung, kidney, bile duct, and melanoma [76]. Malignancy-associated hypercalcemia may also be the cause of acute pancreatitis [77]. The risk of primary pancreatic cancer is significantly increased in patients with hereditary pancreatitis due to genetic mutations; the risk of pancreatic adenocarcinoma in these individuals is as high as 54% by age of 75 years [78].

Acute pancreatitis may also develop in patients undergoing antineoplastic chemotherapy. While in general, less than 2% of acute pancreatitis is drug induced, the development of acute pancreatitis has been infrequently associated with antineoplastic chemotherapy. While it is often impossible to definitively conclude that a particular drug is the etiology of pancreatitis without rechallenge, multiple cases of chemotherapy-induced pancreatitis have been reported, including with capecitabine, paclitaxel, bortezomib, vinorelbine, and ifosfamide. Rechallenge was not attempted in many of the reported cases [79]. Tamoxifen may act through induction of hypertriglyceridemia to induce pancreatitis [80]. It is important to recognize that patients receiving chemotherapy can develop acute pancreatitis independent of their tumor or therapy for malignancy due to common etiologies such as gallstone disease or alcohol.

The management of patients with acute pancreatitis involves administration of aggressive intravenous fluid hydration, analgesia, and bowel rest. Aggressive hydration is defined as 250–500 mL/h (unless cardiovascular or renal indications dictate otherwise), with close observation of urine output and adjustment of hydration as needed, with the goals of decreasing blood urea nitrogen [81]. Early effective management of acute pancreatitis is critical to prevent multiorgan failure (renal failure, hypotension, respiratory compromise, and cardiovascular collapse) that can ensue as a result of the systemic inflammatory response syndrome [82, 83].

Inadequate early hydration may lead to any of these devastating consequences and is one of the most common clinical pitfalls in the management of acute pancreatitis. Furthermore, strict bowel rest (nil per os status, NPO) until patients are pain-free to limit pancreatic stimulation is a hallmark of initial management. The vast majority of patients have a mild course with inpatient admission for <5-7 days. In those cases in which the patient's clinical course dictates prolonged NPO status for more than 3-5 days, supplemental nutrition is indicated, preferably by enteral route with NG or nasojejunal (NJ) feeding. The optimal strategy is usually to allow the patient to consume nutrition per os; however, many patients are unable to meet the metabolic demands via only PO intake, and thus enteral nutritional supplementation should be instituted. In patients with acute pancreatitis, enteral nutrition significantly reduced mortality, multiple organ failure, systemic infections, and the need for operative interventions compared to those who received TPN. This was demonstrated in a meta-analysis of eight randomized controlled trials [84].

Patients who develop complications as a result of acute pancreatitis such as acute pancreatic fluid collections, pancreatic necrosis, or pseudocyst warrant a multidisciplinary discussion with gastroenterologists, including interventional endoscopists and pancreatic surgeons to optimize management of such complications.

Biliary Obstruction

Malignant obstruction of the biliary tree can arise from primary tumors of the bile duct (intrahepatic or extrahepatic cholangiocarcinomas) or from extrinsic compression and/or invasion of the bile duct by pancreatic, ampullary, or duodenal cancers and lymphadenopathy (peripancreatic or portal lymph nodes), or from metastatic spread to the biliary tree or liver. Biliary obstruction manifests as jaundice, acholic stool, dark urine, pruritus, abdominal pain, nausea, and weight loss. Cholangitis may occur as a result of biliary stasis and subsequent infection; however, in general, in the absence of prior biliary intervention or choledocholithiasis, ascending bacterial cholangitis is uncommon in patients with malignant biliary obstruction.

Biliary obstruction is diagnosed by abnormalities in serum bilirubin, alkaline phosphatase, and liver transaminases, as well as imaging (ultrasound, CT, MRI) showing evidence of obstruction such as biliary dilation proximal to the site of obstruction. Once established, it is important to triage the urgency of biliary decompression. Patients with asymptomatic jaundice do not require biliary decompression unless their hyperbilirubinemia interferes with chemotherapy (i.e., some chemotherapeutic regimens require a normal bilirubin). Patients with intolerable jaundice or pruritus or poor nutritional status as a result of hyperbilirubinemia should have elective biliary decompression. Pruritus associated with hyperbilirubinemia can be debilitating and has been managed with antihistamines, corticosteroids, cholestyramine, and other medications with only limited success, thus relief of obstruction is the mainstay of treatment. Those patients with signs and symptoms of acute cholangitis require urgent drainage and intravenous antibiotics.

Biliary decompression can be accomplished by endoscopic (endoscopic retrograde cholangiopancreatography, [ERCP]) (Fig. 25.4), percutaneous (interventional radiology), or surgical means. With the development and advances in endoscopic and percutaneous drainage procedures, surgical decompression is rarely utilized in modern clinical practice.

Endoscopic and percutaneous biliary drainage procedures have their individual risks and benefits. Common risks to both procedures include infection, stent occlusion, or migration. Endoscopic drainage via ERCP is considered relatively noninvasive, highly successful, and well tolerated. However, it carries risks associated with bleeding specifically from the biliary sphincterotomy site, intestinal perforation, and procedure-related pancreatitis (generally 3–10%) [85]. Percutaneous transhepatic cholangiography (PTC) eliminates the potential for acute pancreatitis and intestinal perforation and requires less sedation than ERCP. It is highly successful especially in high-volume centers through, may be challenging in cases where there is no significant intrahepatic biliary dilation to serve as a target. This means biliary drainage may be necessary in the setting of gastroduodenal obstruction, although new endoscopic ultrasound-guided adjunct techniques allow for biliary drainage in this setting (i.e., choledochoduodenostomy). Percutaneous biliary drains can be replaced and exchanged with relative ease. Nevertheless, percutaneous biliary drainage does leave the patient with an external drain which may impact their quality of life, while endoscopic drainage obviates the need for external catheter drainage.

ERCP-placed stents can be plastic or metal (selfexpandable metal stents, SEMSs). The advantage of plastic stents is their low cost and relative ease of removal, if needed, at the time of surgery. Plastic stents, however, have shorter life span due to their smaller diameter (maximum diameter of plastic stent is 12 French, 4 mm) and may not maintain patency long enough to allow for neoadjuvant chemoradiotherapy in cases of pancreatic adenocarcinoma. Up to 55% of patients with ERCP-placed biliary plastic stents for malignant obstruction from pancreatic cancer require additional ERCP intervention for biliary obstruction or cholangitis related to stent occlusion [86]. Biliary SEMS do not adversely affect surgical outcomes and are preferable for more durable stenting, however are more expensive upfront. Decision analysis studies have not shown plastic stents to be more cost effective for pancreatic cancer [87, 88].

While biliary metal stents have been shown to be superior to plastic stents for decompression caused by pancreatic adenocarcinoma by maintaining patency longer, stent occlusion may still develop. In a multicenter study of 241 patients treated with metal stents as part of preoperative protocol, 5.8% of patients developed stent occlusion with a median time to occlusion of 6.6 months (range 1–20 months) [89]. Mechanisms of metal and plastic stent malfunction differ.



Fig. 25.4 Obstructing pancreatic adenocarcinoma. (**a**) A 2.5-cm illdefined lesion in the pancreatic head on CT scan; (**b**) pancreatic head mass (*arrowhead*) resulting in biliary obstruction and dilated common

bile duct (*arrow*) on endoscopic ultrasound; and (**c**) tight fluoroscopic waist within ERCP-placed self-expanding biliary metal stent for obstructing pancreatic head mass

Plastic stent may occlude due to formation of bacterial biofilm and sludge formation, or they may migrate proximally or distally and no longer span the obstructing biliary stricture. There may be tumor overgrowth covering the proximal end of the stent. Covered metal stents may have similar mechanisms for occlusion. Uncovered metal stents are less likely to migrate; however, there may be tumor or benign reactive tissue ingrowth, in addition to biofilm and sludge formation.

Patients with occluded biliary stents or percutaneous catheters may present with recurrence of their original symptoms that led to interventions, as well as jaundice, fever, chills, and abdominal pain. Intermittent obstruction due to sludge and stones is possible. Imaging studies may show biliary dilation. Lack of pneumobilia (an expected finding in patients with patent biliary stents) may be a radiologic clue to biliary obstruction, but it is neither a specific nor sensitive finding.

The acute management of biliary obstruction with cholangitis in patients with malignancy is not different from patients with benign disease and involves management of sepsis with intravenous fluids, broad-spectrum antibiotics, and prompt establishment of biliary drainage via endoscopic or percutaneous methods. Endoscopic management options of occluded stents include removal of the occluded stent and placement of new covered or uncovered metal stent or plastic stent. Percutaneous catheter exchange in patients with prior PTC is the treatment of choice for acute cholangitis.

Management of these complicated patients necessitates a multidisciplinary team with input from medical, radiation, and surgical oncologists, the patient's primary care physician, interventional radiologists, and gastroenterologists to determine the optimal management strategy.

Hepatic Decompensation

Fulminant Hepatic Failure

Fulminant hepatic failure due to malignant infiltration of the liver is rare, but has been reported in case studies in the literature [90, 91]. These case reports have centered on diffuse infiltration by tumor cells (lymphoma and infiltrative carcinoma) rather than numerous hepatic metastases and could not be distinguished on cross-sectional imaging. Though it is a rare complication of metastatic liver disease, it carries a high mortality. Many chemotherapeutic agents may have hepatotoxic manifestations and thus the selection of the appropriate antineoplastic regimen must take into account the patient's baseline liver chemistries and associated comorbidities. Polypharmacy which many oncologic patients are the victims of is also an additional cause of abnormal liver chemistries. Most often, drug induced liver injury is self-

limited (possibly limited to abnormal liver chemistries) and can be reversed with removal of the offending agent; however in rare cases, it may lead to fulminant hepatic failure with a decline in synthetic liver function.

In the setting of fulminant hepatic failure, patients usually present with jaundice, altered mental status or bleeding. The etiology is due to replacement or destruction of hepatocytes with tumor resulting in the compromised liver synthetic function. As a result, there is decreased synthesis of albumin and oncotic proteins which promote the development of ascites, decreased conjugation of bilirubin resulting in jaundice, or obstruction of intrahepatic ducts from tumor resulting in jaundice, bleeding due to elevations in the prothrombin/international normalized ratio (INR) due to derangements in the synthesis of key clotting factors, and eventually decreased ability of the liver to process toxins resulting in encephalopathy. Cerebral edema may develop in patients with acute liver failure leading to increased intracranial pressure and risk of subsequent herniation. The early recognition of acute liver failure is critical given its overall dismal prognosis if left untreated. In cases in which the etiology is nonmalignant, lifesaving measures such as orthotopic liver transplantation should be considered. Clearly, when malignancy is the underlying etiology, recognition of hepatic decompensation is a sign of terminal prognosis.

Laboratory evaluation initially should include liver enzymes including AST, ALT, alkaline phosphatase, GGT, total and direct bilirubin, albumin, prothrombin time/INR, serum chemistries, ammonia level, and viral serological tests. Cross-sectional imaging may show diffuse hepatic involvement by tumor resulting in loss of hepatic reserve (Fig. 25.5). It is important to exclude a secondary etiology to liver dysfunction that may be at play and cause the sudden



Fig. 25.5 Innumerable intrahepatic masses due to metastatic colorectal cancer resulting in jaundice and compromised synthetic liver function. (Courtesy of Alice Fung, MD, Oregon Health & Science University)

imbalance in liver synthetic function. Thus, formal hepatology consultation is appropriate. Additionally, it is important to exclude extrahepatic biliary obstruction as an underlying etiology for jaundice as therapy for biliary obstruction may lower the patient's bilirubin enough to allow for further oncologic treatment. In this vein, cross-sectional imaging (CAT scan of the abdomen or preferably MRI/MRCP) and consultation with gastroenterology and interventional radiology may benefit the patient. In general, progressive malignant liver dysfunction leads to hospice. Nonetheless, it is important to exclude reversible contributing illness to a patient's declining hepatic function.

Ascites

Ascites, defined as free intra-abdominal fluid accumulation, may be a result of malignancy with and without liver involvement. Ascites may be detected due to the presence of bloating, abdominal distention, and dullness to abdominal percussion. Abdominal ultrasound may be required to determine with certainty that fluid is present within the abdomen. Breast, lung, colon, and pancreatic primary malignancies are most commonly complicated by ascites [92]. In addition, significant intrahepatic tumor burden may result in liver dysfunction and subsequent ascites.

While there are many possible etiologies of ascites including cirrhosis, Budd-Chiari syndrome, pancreatitis, congestive heart failure, etc., we will focus on malignancy-related ascites. The serum-ascites albumin gradient (SAAG) has been used to categorize ascites; it is able to differentiate portal hypertensive ascites vs. nonportal hypertension-related ascites with a diagnostic accuracy of 97% [93, 94]. In a patient with new ascites, a diagnostic paracentesis is performed in order to calculate this gradient and help identify the etiology. The pathophysiology of malignant ascites is not related to portal hypertension, and thus these patients have a SAAG <1.1 g/dL. There are several pathophysiologic mechanisms of malignant ascites including lymphatic obstruction by lymphoma or mass, low serum oncotic pressure due to various causes (including massive liver metastases with liver dysfunction), and exudation of proteinaceous fluid from tumor cells that line the peritoneum in patients with peritoneal carcinomatosis [92].

Unlike portal hypertension-related ascites in which the underlying pathophysiology involves a complex interplay of splanchnic and systemic vascular resistance coupled with sodium and water retention which in general responds to dietary sodium restriction, diuretics (such as furosemide or spironolactone), or intravascular volume expansion (intravenous albumin), the mainstay of the management of malignant ascites is large-volume paracentesis for symptomatic relief. This can be arranged as an outpatient via ultrasound guidance with interventional radiology or with gastroenterology. The timing and frequency of paracenteses are dictated by patient symptoms. In general, peritoneal catheters are avoided in these patients given the potential for loss of protein and hydration as well as potential to seed the catheter track. In patients with malignancy ascites related to ovarian cancer, tumor debulking and chemotherapy may be effective in ascites management (Sugarbaker technique).

Urgent Issues Related to Enteral Feeding Devices

Many patients with cancer battle with issues related to malnutrition. This is due to the prolonged negative balance of protein and energy below metabolic requirements that result from the tumor itself or intensity of treatments. As such, cancer patients may require supplemental enteral feeding via nasoenteral tubes, gastrostomy, or jejunostomy feeding devices. The type of enteral access is usually decided upon based on the length of anticipated need. Nasoenteric tubes are commonly used and have the benefit of relative ease of placement. Such tubes are utilized when the feeding is predicted to be approximately less than 30–45 days. Dislodgement of nasoenteric tubes is usually not of major concern as they are easily replaced and dislodgement can be prevented by the use of a nasal bridle device [95–97].

When long-term enteral access devices are needed, gastrostomy or jejunostomy tubes are the most common methods. Such devices may be placed by endoscopic (gastroenterology), percutaneous (interventional radiology), or surgical means. Once placed, it is recommended that these devices should not be removed for at least 6 weeks to allow the enterocutaneous fistula to mature prior to intentional removal. The patient should be made aware that once removed, the track can take up to 2 weeks to close and to expect some leakage during this time frame.

Enteral access replacement may be urgently needed in cases of unintentional dislodgement in order to prevent closure of the enterocutaneous fistula. If dislodgement occurs within 14 days of insertion, this track may not be mature, and "blind" reinsertion of a tube via the fistula should not be attempted to avoid erroneous placement of the enteral device into the peritoneal cavity [98]. The patient should be advised to call the provider who initially inserted the enteral device to arrange for reinsertion or present to the emergency department. In cases where the track is mature (>6 weeks from placement) and the device is dislodged, a temporary tube such as a Foley balloon catheter can be inserted in the track to maintain patency until the appropriate replacement device can be inserted and secured. This can be done at the bedside in any emergency. By directly addressing the dislodged tube in a timely manner, the clinician may be able to avoid the need for an endoscopic, interventional radiology, or surgical replacement procedure.

Summary

GI-related oncologic emergencies represent a small proportion of all oncologic emergencies and those herein discussed include GI bleeding, luminal and biliary obstruction, acute pancreatitis, hepatic decompensation, and dislodgement of enteral devices. These disorders require early recognition allowing clinicians, patients, and families to be aware of these possibilities and aid in early identification of such emergencies. A multidisciplinary approach to such patients and conditions, including consultation with oncologic, radiation oncologic, surgery, primary care, and gastroenterology to assist in management (which may include a therapeutic GI procedure), is likely to lead to optimal patient outcomes.

References

- Marmo R, Del Piano M, Rotondano G, Koch M, Bianco MA, Zambelli A, PNED 1 Investigators; PNED 2 Investigators, Prometeo Investigators, et al. Mortality from nonulcer bleeding is similar to that of ulcer bleeding in high-risk patients with nonvariceal hemorrhage: a prospective database study in Italy. Gastrointest Endosc. 2012;75(2):263–72, 272, e1.
- Hwang JH, Fisher DA, Ben-Menachem T, Chandrasekhara V, Chathadi K, Decker GA, Standards of Practice Committee of the American Society for Gastrointestinal Endoscopy, et al. The role of endoscopy in the management of acute non-variceal upper GI bleeding. Gastrointest Endosc. 2012;75(6):1132–8.
- Samuel R, Bilal M, Tayyem O, Guturu P. Evaluation and management of non-variceal upper gastrointestinal bleeding. Dis Mon. 2018;64(7):333–43.
- Ofosu A, Ramai D, Latson W, Adler DG. Endoscopic management of bleeding gastrointestinal tumors. Ann Gastroenterol. 2019;32(4):346–51.
- Boskoski I, Familiari P, Costamagna G. New and emerging endoscopic therapies for gastrointestinal bleeding. Curr Opin Gastroenterol. 2014;30(5):439–43.
- Schatz RA, Rockey DC. Gastrointestinal bleeding due to gastrointestinal tract malignancy: natural history, management, and outcomes. Dig Dis Sci. 2017;62(2):491–501.
- ASGE Technology Committee, Parsi MA, Schulman AR, Aslanian HR, Bhutani MS, Krishnan K, Lichtenstein DR, ASGE Technology Committee Chair, et al. Devices for endoscopic hemostasis of nonvariceal GI bleeding (with videos). VideoGIE. 2019;4(7):285–99.
- Savides TJ, Jensen DM, Cohen J, Randall GM, Kovacs TO, Pelayo E, et al. Severe upper gastrointestinal tumor bleeding: endoscopic findings, treatment, and outcome. Endoscopy. 1996;28(2):244–8.
- Schwesinger WH, Chumley DL. Laser palliation for gastrointestinal malignancy. Am Surg. 1988;54(2):100–4.

- Suzuki H, Miho O, Watanabe Y, Kohyama M, Nagao F. Endoscopic laser therapy in the curative and palliative treatment of upper gastrointestinal cancer. World J Surg. 1989;13(2):158164.
- Loftus EV, Alexander GL, Ahlquist DA, Balm RK. Endoscopic treatment of major bleeding from advanced gastroduodenal malignant lesions. Mayo Clin Proc. 1994;69(8):736–40.
- Eckhauser ML. Laser therapy of gastrointestinal tumors. World J Surg. 1992;16(6):1054–9.
- Mathus-Vliegen EM, Tytgat GN. Analysis of failures and complications of neodymium: YAG laser photocoagulation in gastrointestinal tract tumors. A retrospective survey of 18 years' experience. Endoscopy. 1990;22(1):17–23.
- Kanai M, Hamada A, Endo Y, Takeda Y, Yamakawa M, Nishikawa H, Torii A. Efficacy of argon plasma coagulation in nonvariceal upper gastrointestinal bleeding. Endoscopy. 2004;36(12):1085–8.
- Wahab PJ, Mulder CJ, den Hartog G, Thies JE. Argon plasma coagulation in flexible gastrointestinal endoscopy: pilot experiences. Endoscopy. 1997;29(3):176–81.
- Akhtar K, Byrne JP, Bancewicz J, Attwood SE. Argon beam plasma coagulation in the management of cancers of the esophagus and stomach. Surg Endosc. 2000;14(12):1127–30.
- ASGE Technology Committee, Parsi MA, Trindade AJ, Bhutani MS, Melson J, Navaneethan U, Thosani N, et al. Cryotherapy in gastrointestinal endoscopy. VideoGIE. 2017;2(5):89–95.
- Shaheen NJ, Peery AF, Overholt BF, Lightdale CJ, Chak A, Wang KK, AIM Dysplasia Investigators, et al. Biopsy depth after radiofrequency ablation of dysplastic Barrett's esophagus. Gastrointest Endosc. 2010;72(3):490–496.e1.
- Shah MB, Schnoll-Sussman F. Novel use of cryotherapy to control bleeding in advanced esophageal cancer. Endoscopy. 2010;42(Suppl 2):E46.
- Babiuc RD, Purcarea M, Sadagurschi R, Negreanu L. Use of Hemospray in the treatment of patients with acute UGIB – short review. J Med Life. 2013;6(2):117–9.
- Bustamante-Balen M, Plume G. Role of hemostatic powders in the endoscopic management of gastrointestinal bleeding. World J Gastrointest Pathophysiol. 2014;5(3):284–92.
- Chen YI, Barkun AN, Soulellis C, Mayrand S, Ghali P. Use of the endoscopically applied hemostatic powder TC-325 in cancerrelated upper GI hemorrhage: preliminary experience (with video). Gastrointest Endosc. 2012;75(6):1278–81.
- 23. Leblanc S, Vienne A, Dhooge M, Coriat R, Chaussade S, Prat F. Early experience with a novel hemostatic powder used to treat upper GI bleeding related to malignancies or after therapeutic interventions (with videos). Gastrointest Endosc. 2013;78(1):169–75.
- Ozaslan E, Purnak T, Yildiz A, Akar T, Avcioglu U, Haznedaroglu IC. The effect of Ankaferd blood stopper on severe radiation colitis. Endoscopy. 2009;41(Suppl 2):E321–2.
- 25. Kurt M, Akdogan M, Onal IK, Kekilli M, Arhan M, Shorbagi A, et al. Endoscopic topical application of Ankaferd Blood Stopper for neoplastic gastrointestinal bleeding: a retrospective analysis. Dig Liver Dis. 2010;42(3):196–9.
- Chen YI, Wyse J, Lu Y, Martel M, Barkun AN. TC-325 hemostatic powder versus current standard of care in managing malignant GI bleeding: a pilot randomized clinical trial. Gastrointest Endosc. 2020;91(2):321–e328. e1.
- Cheng AW, Chiu PW, Chan PC, Lam SH. Endoscopic hemostasis for bleeding gastric stromal tumors by application of hemoclip. J Laparoendosc Adv Surg Tech A. 2004;14(3):169–71.
- Heller SJ, Tokar JL, Nguyen MT, Haluszka O, Weinberg DS. Management of bleeding GI tumors. Gastrointest Endosc. 2010;72(4):817–24.
- Chaw CL, Niblock PG, Chaw CS, Adamson DJ. The role of palliative radiotherapy for haemostasis in unresectable gastric cancer: a single-institution experience. Ecancermedicalscience. 2014;8:384.

- Blackshaw GR, Stephens MR, Lewis WG, Paris HJ, Barry JD, Edwards P, Allison MC. Prognostic significance of acute presentation with emergency complications of gastric cancer. Gastric Cancer. 2004;7(2):91–6.
- Vasas P, Wiggins T, Chaudry A, Bryant C, Hughes FS. Emergency presentation of the gastric cancer; prognosis and implications for service planning. World J Emerg Surg. 2012;7(1):31.
- Saidi RF, ReMine SG, Dudrick PS, Hanna NN. Is there a role for palliative gastrectomy in patients with stage IV gastric cancer? World J Surg. 2006;30(1):21–2.
- Barr H, Greenall MJ. Carcinoma of the stomach. In: Morris PJ, Wood WC, editors. Oxford textbook of surgery. 2nd ed. Oxford: Oxford University Press; 2000. p. 1313–28.
- 34. Lupaşcu C, Andronic D, Ursulescu C, Vasiluţă C, Raileanu G, Georgescu S, et al. Palliative gastrectomy in patients with stage IV gastric cancer--our recent experience. Chirurgia (Bucur). 2010;105(4):473–6.
- 35. Schultheiss TE, Lee WR, Hunt MA, Hanlon AL, Peter RS, Hanks GE. Late GI and GU complications in the treatment of prostate cancer. Int J Radiat Oncol Biol Phys. 1997;37(1):3–11.
- Beard CJ, Propert KJ, Rieker PP, Clark JA, Kaplan I, Kantoff PW, Talcott JA. Complications after treatment with external-beam irradiation in early-stage prostate cancer patients: a prospective multiinstitutional outcomes study. J Clin Oncol. 1997;15(1):223–9.
- Chon BH, Loeffler JS. The effect of nonmalignant systemic disease on tolerance to radiation therapy. Oncologist. 2002;7(2):136–43.
- Tabaja L, Sidani SM. Management of radiation proctitis. Dig Dis Sci. 2018;63(9):2180–8.
- Kochhar R, Sriram PV, Sharma SC, Goel RC, Patel F. Natural history of late radiation proctosigmoiditis treated with topical sucralfate suspension. Dig Dis Sci. 1999;44(5):973–8.
- Sasai T, Hiraishi H, Suzuki Y, Masuyama H, Ishida M, Terano A. Treatment of chronic post-radiation proctitis with oral administration of sucralfate. Am J Gastroenterol. 1998;93(9):1593–5.
- Chun M, Kang S, Kil HJ, Oh YT, Sohn JH, Ryu HS. Rectal bleeding and its management after irradiation for uterine cervical cancer. Int J Radiat Oncol Biol Phys. 2004;58(1):98–105.
- Taylor JG, DiSario JA, Buchi KN. Argon laser therapy for hemorrhagic radiation proctitis: long-term results. Gastrointest Endosc. 1993;39(5):641–4.
- Sarin A, Safar B. Management of radiation proctitis. Gastroenterol Clin N Am. 2013;42(4):913–25.
- 44. Karamanolis G, Triantafyllou K, Tsiamoulos Z, Polymeros D, Kalli T, Misailidis N, Ladas SD. Argon plasma coagulation has a longlasting therapeutic effect in patients with chronic radiation proctitis. Endoscopy. 2009;41(6):529–31.
- Karamanolis G, Psatha P, Triantafyllou K. Endoscopic treatments for chronic radiation proctitis. World J Gastrointest Endosc. 2013;5(7):308–12.
- 46. Eddi R, Depasquale JR. Radiofrequency ablation for the treatment of radiation proctitis: a case report and review of literature. Ther Adv Gastroenterol. 2013;6(1):69–76.
- 47. Zhou C, Adler DC, Becker L, Chen Y, Tsai TH, Figueiredo M, et al. Effective treatment of chronic radiation proctitis using radiofrequency ablation. Ther Adv Gastroenterol. 2009;2(3):149–56.
- Moawad FJ, Horwhat JD. Cryotherapy for treatment of radiation proctitis. Gastrointest Endosc. 2014;79(2):209–10.
- Moawad FJ, Maydonovitch CL, Horwhat JD. Efficacy of cryospray ablation for the treatment of chronic radiation proctitis in a pilot study. Dig Endosc. 2013;25(2):174–9.
- Oscarsson N, Arnell P, Lodding P, Ricksten SE, Seeman-Lodding H. Hyperbaric oxygen treatment in radiation-induced cystitis and proctitis: a prospective cohort study on patient-perceived quality of recovery. Int J Radiat Oncol Biol Phys. 2013;87(4):670–5.
- Bosscher MR, van Leeuwen BL, Hoekstra HJ. Surgical emergencies in oncology. Cancer Treat Rev. 2014;40(8):1028–36.

- 52. Sharma P, Kozarek R, Practice Parameters Committee of American College of Gastroenterology. Role of esophageal stents in benign and malignant diseases. Am J Gastroenterol. 2010;105(2):258–73; quiz 274.
- 53. Siddiqui AA, Sarkar A, Beltz S, Lewis J, Loren D, Kowalski T, et al. Placement of fully covered self-expandable metal stents in patients with locally advanced esophageal cancer before neoadjuvant therapy. Gastrointest Endosc. 2012;76(1):44–51.
- 54. Ding NS, Alexander S, Swan MP, Hair C, Wilson P, Clarebrough E, Devonshire D. Gastroduodenal outlet obstruction and palliative self-expandable metal stenting: a dual-centre experience. J Oncol. 2013;2013:167851.
- Boskoski I, Tringali A, Familiari P, Mutignani M, Costamagna G. Self-expandable metallic stents for malignant gastric outlet obstruction. Adv Ther. 2010;27(10):691–703.
- 56. Larssen L, Hauge T, Medhus AW. Stent treatment of malignant gastric outlet obstruction: the effect on rate of gastric emptying, symptoms, and survival. Surg Endosc. 2012;26(10):2955–60.
- 57. Larssen L, Medhus AW, Körner H, Glomsaker T, Søberg T, Gleditsch D, et al. Long-term outcome of palliative treatment with self-expanding metal stents for malignant obstructions of the GI tract. Scand J Gastroenterol. 2012;47(12):1505–14.
- Tyberg A, Perez-Miranda M, Zerbo S, Baron TH, Kahaleh M. Endoscopic ultrasound-guided gastrojejunostomy: a novel technique. Endoscopy. 2017;49(10):E252–3.
- Perez-Miranda M, Tyberg A, Poletto D, Toscano E, Gaidhane M, Desai AP, et al. EUS-guided gastrojejunostomy versus laparoscopic gastrojejunostomy: an international collaborative study. J Clin Gastroenterol. 2017;51(10):896–9.
- 60. Khashab MA, Bukhari M, Baron TH, Nieto J, El Zein M, Chen YI, et al. International multicenter comparative trial of endoscopic ultrasonography-guided gastroenterostomy versus surgical gastrojejunostomy for the treatment of malignant gastric outlet obstruction. Endosc Int Open. 2017;5(4):E275–81.
- Kaw M, Singh S, Gagneja H. Clinical outcome of simultaneous self-expandable metal stents for palliation of malignant biliary and duodenal obstruction. Surg Endosc. 2003;17(3):457–61.
- 62. Moon JH, Choi HJ, Ko BM, Koo HC, Hong SJ, Cheon YK, et al. Combined endoscopic stent-in-stent placement for malignant biliary and duodenal obstruction by using a new duodenal metal stent (with videos). Gastrointest Endosc. 2009;70(4):772–7.
- 63. García-Cano J, González-Huix F, Juzgado D, Igea F, Pérez-Miranda M, López-Rosés L, et al. Use of self-expanding metal stents to treat malignant colorectal obstruction in general endoscopic practice (with videos). Gastrointest Endosc. 2006;64(6):914–20.
- 64. van Hooft JE, Fockens P, Marinelli AW, Timmer R, van Berkel AM, Bossuyt PM, Bemelman WA, Dutch Colorectal Stent Group. Early closure of a multicenter randomized clinical trial of endoscopic stenting versus surgery for stage IV left-sided colorectal cancer. Endoscopy. 2008;40(3):184–91.
- 65. Frago R, Kreisler E, Biondo S, Alba E, Domínguez J, Golda T, et al. Complicaciones del tratamiento de la oclusión del colon distal con prótesis endoluminales [Complications of distal intestinal occlusion treatment with endoluminal implants]. Cir Esp. 2011;89(7):448– 55. (*Article in Spanish*).
- 66. Arezzo A, Passera R, Lo Secco G, Verra M, Bonino MA, Targarona E, Morino M. Stent as bridge to surgery for left-sided malignant colonic obstruction reduces adverse events and stoma rate compared with emergency surgery: results of a systematic review and meta-analysis of randomized controlled trials. Gastrointest Endosc. 2017;86(3):416–26.
- 67. Xinopoulos D, Dimitroulopoulos D, Theodosopoulos T, Tsamakidis K, Bitsakou G, Plataniotis G, et al. Stenting or stoma creation for patients with inoperable malignant colonic obstructions? Results of a study and cost-effectiveness analysis. Surg Endosc. 2004;18(3):421–6.

- 68. Fernández-Esparrach G, Bordas JM, Giráldez MD, Ginès A, Pellisé M, Sendino O, et al. Severe complications limit long-term clinical success of self-expanding metal stents in patients with obstructive colorectal cancer. Am J Gastroenterol. 2010;105(5):1087–93.
- Frago R, Ramirez E, Millan M, Kreisler E, del Valle E, Biondo S. Current management of acute malignant large bowel obstruction: a systematic review. Am J Surg. 2014;207(1):127–38.
- Lin A, Feller ER. Pancreatic carcinoma as a cause of unexplained pancreatitis: report of ten cases. Ann Intern Med. 1990;113(2):166–7.
- Munigala S, Kanwal F, Xian H, Scherrer JF, Agarwal B. Increased risk of pancreatic adenocarcinoma after acute pancreatitis. Clin Gastroenterol Hepatol. 2014;12(7):1143–50. e1141.
- Tummala P, Tariq SH, Chibnall JT, Agarwal B. Clinical predictors of pancreatic carcinoma causing acute pancreatitis. Pancreas. 2013;42(1):108–13.
- Tejedor Bravo M, Justo LM, Lasala JP, Moreira Vicente VF, Ruiz AC, Scapa ML. Acute pancreatitis secondary to neuroendocrine pancreatic tumors: report of 3 cases and literature review. Pancreas. 2012;41(3):485–9.
- 74. Vasiliadisl K, Papavasiliou C, Pervana S, Nikopoulos K, Makridis C. Acute pancreatitis as the initial manifestation of an adenocarcinoma of the major duodenal papilla in a patient with familial adenomatous polyposis syndrome: a case report and literature review. Acta Chir Belg. 2013;113(6):463–7.
- Stewart KC, Dickout WJ, Urschel JD. Metastasis-induced acute pancreatitis as the initial manifestation of bronchogenic carcinoma. Chest. 1993;104(1):98–100.
- Sobesky R, Duclos-Vallée JC, Prat F, Pelletier G, Encaoua R, Boige V, et al. Acute pancreatitis revealing diffuse infiltration of the pancreas by melanoma. Pancreas. 1997;15(2):213–5.
- McIntosh J, Lauer J, Gunatilake R, Knudtson E. Multiple myeloma presenting as hypercalcemic pancreatitis during pregnancy. Obstet Gynecol. 2014;124(2 Pt 2 Suppl 1):461–3.
- Rebours V, Boutron-Ruault MC, Schnee M, Férec C, Le Maréchal C, Hentic O, et al. The natural history of hereditary pancreatitis: a national series. Gut. 2009;58(1):97–103.
- Lu L, Lou Y, Tan H. Chemotherapy-induced fulminant acute pancreatitis in pancreatic carcinoma: a case report. Oncol Lett. 2014;8(3):1143–6.
- Davila M, Bresalier RS. Gastrointestinal complications of oncologic therapy. Nat Clin Pract Gastroenterol Hepatol. 2008;5(12):682–96.
- Tenner S, Baillie J, DeWitt J, Vege SS, American College of Gastroenterology. American College of Gastroenterology guideline: management of acute pancreatitis. Am J Gastroenterol. 2013;108(9):1400–15; 1416. Erratum in: Am J Gastroenterol. 2014;109(2):302.
- 82. Wu BU, Banks PA. Clinical management of patients with acute pancreatitis. Gastroenterology. 2013;144(6):1272–81.
- Fisher JM, Gardner TB. The "golden hours" of management in acute pancreatitis. Am J Gastroenterol. 2012;107(8):1146–50.
- Al-Omran M, Albalawi ZH, Tashkandi MF, Al-Ansary LA. Enteral versus parenteral nutrition for acute pancreatitis. Cochrane Database Syst Rev. 2010;(1):CD002837.

- 85. ASGE Standards of Practice Committee, Chandrasekhara V, Khashab MA, Muthusamy VR, Acosta RD, Agrawal D, Bruining DH, et al. Adverse events associated with ERCP. Gastrointest Endosc. 2017;85(1):32–47.
- Boulay BR, Gardner TB, Gordon SR. Occlusion rate and complications of plastic biliary stent placement in patients undergoing neoadjuvant chemoradiotherapy for pancreatic cancer with malignant biliary obstruction. J Clin Gastroenterol. 2010;44(6):452–5.
- 87. Gardner TB, Spangler CC, Byanova KL, Ripple GH, Rockacy MJ, Levenick JM, et al. Cost-effectiveness and clinical efficacy of biliary stents in patients undergoing neoadjuvant therapy for pancreatic adenocarcinoma in a randomized controlled trial. Gastrointest Endosc. 2016;84(3):460–6.
- Chen VK, Arguedas MR, Baron TH. Expandable metal biliary stents before pancreaticoduodenectomy for pancreatic cancer: a Monte-Carlo decision analysis. Clin Gastroenterol Hepatol. 2005;3(12):1229–37.
- 89. Siddiqui AA, Mehendiratta V, Loren D, Kowalski T, Fang J, Hilden K, Adler DG. Self-expanding metal stents (SEMS) for preoperative biliary decompression in patients with resectable and borderline-resectable pancreatic cancer: outcomes in 241 patients. Dig Dis Sci. 2013;58(6):1744–50.
- 90. Alexopoulou A, Koskinas J, Deutsch M, Delladetsima J, Kountouras D, Dourakis SP. Acute liver failure as the initial manifestation of hepatic infiltration by a solid tumor: report of 5 cases and review of the literature. Tumori. 2006;92(4):354–7.
- Rowbotham D, Wendon J, Williams R. Acute liver failure secondary to hepatic infiltration: a single centre experience of 18 cases. Gut. 1998;42(4):576–80.
- Runyon BA, Hoefs JC, Morgan TR. Ascitic fluid analysis in malignancy-related ascites. Hepatology. 1988;8(5):1104–9.
- Hoefs JC. Serum protein concentration and portal pressure determine the ascitic fluid protein concentration in patients with chronic liver disease. J Lab Clin Med. 1983;102(2):260–73.
- 94. Runyon BA, Montano AA, Akriviadis EA, Antillon MR, Irving MA, McHutchison JG. The serum-ascites albumin gradient is superior to the exudate-transudate concept in the differential diagnosis of ascites. Ann Intern Med. 1992;117(3):215–20.
- 95. Popovich MJ, Lockrem JD, Zivot JB. Nasal bridle revisited: an improvement in the technique to prevent unintentional removal of small-bore nasoenteric feeding tubes. Crit Care Med. 1996;24(3):429–31.
- 96. Seder CW, Stockdale W, Hale L, Janczyk RJ. Nasal bridling decreases feeding tube dislodgment and may increase caloric intake in the surgical intensive care unit: a randomized, controlled trial. Crit Care Med. 2010;38(3):797–801.
- Seder CW, Janczyk R. The routine bridling of nasojejunal tubes is a safe and effective method of reducing dislodgement in the intensive care unit. Nutr Clin Pract. 2008;23(6):651–4.
- Taheri MR, Singh H, Duerksen DR. Peritonitis after gastrostomy tube replacement: a case series and review of literature. JPEN J Parenter Enteral Nutr. 2011;35(1):56–60.

The Acute Abdomen

Linda Ferrari and Alessandro Fichera

Case Study

History present illness: A 57-year-old woman presenting to the emergency department for persistent left upper quadrant pain, nausea, and weight loss. Pain is constant in nature over the past 5 days. She has been having regular bowel movements, formed stools without blood or mucus.

Laboratory values: Hb 107 g/L (120–150 g/L), CRP 272 mg/L (0–4 mg/L).

Abdominal radiograph: No features of intestinal obstruction or perforation

CT chest, abdomen, and pelvis: Highly suspicious wall thickening involving the distal transverse colon/splenic flexure with extramural involvement of the left colic and inferior mesenteric vein. Regional lymph nodes are enlarged. There is an associated collection measuring approximately $4.8 \times$ 4.1 cm which is suggestive of focal tumor perforation. There is surrounding inflammatory changes and it is difficult to exclude associated peritoneal involvement. Such collection is abutting the gastric wall with edematous wall thickening and a likely perforation into the gastric lumen. No definite distant metastatic disease.

Past medical history: Seven years before, patient had right breast invasive cancer. She underwent mastectomy with implant-based reconstruction, axillary clearance, chemoradiation therapy, and adjuvant endocrine treatment with Letrozole (aromatase inhibitor).

L. Ferrari

A. Fichera (🖂)

Hospital course: The patient underwent upper endoscopy which showed a lesion within the stomach, biopsy showing inflammatory tissue and a colonoscopy which showed a transverse colon lesion consistent with invasive adenocarcinoma. The patient was then discharged awaiting colorectal multidisciplinary team discussion and readmitted 2 days later for nausea and inability to tolerate oral intake. An extended right hemicolectomy and en-block partial gastrectomy was recommended and performed as a combined procedure involving an upper and lower gastrointestinal surgeons. The anastomosis was protected with a diverting loop ileostomy.

Key message: Patients with cancer involvement of multiple organs should be management in the context of a multidisciplinary team to optimize outcome.

Small Bowel Obstruction

Small bowel obstruction (SBO) is a common condition responsible for over 300,000 annual hospital admissions in the United States [1], associated with 10% of mortality rate and significant morbidities among survivors [2]. Small bowel obstruction occurs when the flow of contents through the intestines is interrupted. Simple SBO may lead to intestinal dilation, hypersecretion, bacterial overgrowth, and vascular impairment. Closed loop obstruction may immediately compromise mesenteric blood supply leading to ischemia, necrosis, and perforation with subsequent peritonitis.

Patients with SBO are often unable to tolerate oral intake for several days and they should be considered as patients having acute intestinal failure, which has been defined by the European Society of Coloproctology as "the reduction of function below the minimum necessary for the absorption of macronutrients and/or fluids and electrolytes, such that intravenous supplementation is required to maintain health and/ or growth" [3].



[©] Springer Nature Switzerland AG 2021

K. H. Todd et al. (eds.), Oncologic Emergency Medicine, https://doi.org/10.1007/978-3-030-67123-5_26

Colorectal Department, Guy's and St. Thomas' NHS Foundation Trust, London, UK

Division of Colon and Rectal Surgery, Department of Surgery, Baylor University Medical Center, Dallas, TX, USA e-mail: alessandro.fichera@BSWHealth.org

The most frequent cause of small bowel obstruction is abdominal adhesions following previous surgeries which account for approximately 60–70%, followed by hernia in approximately 15% of the cases, including incisional, inguinal, and parastomal [4]. Cancer has been estimated to be the third cause of small bowel obstruction, with approximately 8–10% cases. In the National Audit for Small Bowel Obstruction (NASBO) [5], 205 (8.4%) out of 2431 patients had cancer as the primary cause of small bowel obstruction. Of those, disseminated intra-abdominal malignancies are the most common cause responsible for 70% of cancer cases, followed by right colon cancer (24.4%), while primary tumors of the small bowel represent only 4.9%.

Malignant bowel obstruction (MaSBO) has been defined "as definitive diagnosis of intestinal obstruction distal to the ligament of Treitz and the presence of incurable intra-abdominal malignancy or extraabdominal primary (melanoma, breast, and lung) with known intraperitoneal metastases" [6]. MaSBO is a life-threatening complication of intraabdominal cancer, with an incidence between 5% and 51% in patients with ovarian carcinoma and between 10% and 28% in patients with gastrointestinal malignancies [7]. Different mechanisms are responsible for the occlusion, such as a cancer obstructing the lumen, impaired intestinal motility, medication-related (i.e., opioid analgesics) and radiation therapy complications or sequelae [8].

Obstruction in a patient with previous history of cancer or with a recent diagnosis may also be secondary to benign causes including fibrosis or stricture, postoperative adhesions or ileus, medication-related dysmotility or treatmentrelated edema. Generally, MaSBO of the small bowel is more likely to be multifocal and extrinsic; the most common cancers leading to this are metastatic ovarian and colorectal adenocarcinoma with peritoneal implants, direct invasion, or carcinomatosis [8]. Other causes of SBO may include an obstructing mass which can be extrinsic and compressing the bowel (e.g., carcinomatosis with implants) or less frequently intrinsic (e.g., small bowel adenocarcinoma, lymphoma).

Goals of therapy for SBO include relieving symptoms of nausea, vomiting, and pain, allowing oral intake. These goals, in the case of MaSBO or primary small bowel cancer, should be weighed against the high morbidity and mortality rate in this cohort of patients [5, 9–11] and palliative care considerations should be entertained. The goals for treatment should be individualized for each patient especially for those facing end-of-life decisions.

A palliative, nonoperative approach should be offered for patients with poor performance status, multiple sites of obstruction, carcinomatosis, or ascites. Recurrent obstruction may be as high as 50% in these patients. Patients not suitable for surgical intervention or who refuse surgical care, should receive pharmacological therapy with the aim to reduce bowel edema and inflammation and improve symptoms such as pain, nausea, and vomiting. Combination of medications including antiemetics, drugs to reduce gastrointestinal secretions, anticholinergic medications to reduce peristalsis and secretions, opioids for pain control, and corticosteroids to decrease the tumor-associated edema (and act as antiemetics) are also widely utilized. Patients who are not operative candidates and do not respond to medical management may still benefit from endoscopic stenting or gastrostomy tube placement. The goals of medical management should be aimed at limiting pain and improving quality of life. In cases of overwhelming cancer burden or transition to comfort care, artificial nutrition and hydration may be discontinued. The focus of care should be shifted to symptom relief and hospice care.

Clinical Presentation and Initial Assessment

Patients often initially present with nausea, emesis, and abdominal pain. Depending on the location of the obstruction as well as whether the obstruction is partial or complete, patients may or may not initially present with abdominal bloating and distention. A patient with a proximal obstruction or tumor encasing the proximal small bowel may not present with distention, but rather severe nausea and emesis, often projectile in nature.

Clinical evaluation of SBO should begin with identifying the where, when and how of the patient symptoms—where the obstruction is (proximal or mid-small bowel versus large bowel), when symptoms started and their duration, and how complete (or partial) obstruction appears to be. A clinical estimate of the level of obstruction may be made from the frequency of the pain (short intervals tend to correlate with SBO versus longer intervals typical of large bowel obstruction) and distention. The duration of symptoms should include any gradual or sudden changes in bowel habits, flatus, or bowel movements. Complete vs. partial obstruction may be inferred from current bowel activity.

Pain from SBO tends to be periumbilical and colicky initially. The colicky nature of pain is secondary to compensatory increased intestinal motility that initially occurs to counter the obstruction. However, such intestinal activity eventually subsides, and there are fewer contractions especially with the use of opioid analgesics. As hypoactivity ensues and intraluminal pressure increases, microvascular perfusion may be compromised. Obstipation may follow with complete bowel obstruction.

Differential diagnosis should include adhesive disease and hernias (including abdominal wall and internal hernias) as well as constipation, volvulus, stricture, and ileus, due to chronic narcotic use, are not uncommon scenario in cancer patients. Additionally, neurogenic and metabolic etiologies should be entertained. History of surgeries, radiation, and any anticholinergic medications should be elicited.

A complete physical exam includes evaluation for hernias and a rectal exam to evaluate for the presence of stool in the rectal vault, obstructing rectal lesions, fecal impaction or occult bleeding. If the patient has an ileostomy or colostomy, the stoma should also be digitally examined to ensure no obstruction at the level of the fascia. Abdominal distention may lead to dyspnea. Jaundice should raise concern for involvement of the hepatobiliary tree including lymphadenopathy at the porta hepatis or liver metastases.

The most important task in the emergency department (ED) assessment is to determine whether the patient requires immediate operative intervention. Indications for immediate operative intervention include incarcerated abdominal hernia, frank peritonitis, sepsis, or other findings concerning ischemic bowel disease. Fever, tachycardia, increased fluid requirement, leukocytosis, and certainly signs of sepsis should raise concern for abscess or perforation in the setting of obstruction.

Diagnostic Workup

The diagnosis of SBO is often based on clinical findings. Predictors of small bowel obstruction are: previous history of abdominal surgery, obstipation, and abnormal distension on examination [12]. Minimum laboratory tests include blood count, lactate level, and a metabolic panel.

Abdominal X-ray should be the first imaging modality for most patients with suspected SBO coming from the emergency department due to widespread availability, easy interpretation, and low cost. A single upright chest X-ray may exclude subdiaphragmatic free air, and it is a simple and quick test to determine the need for an emergency exploration, especially in patients whose exam is unreliable (unresponsive, immunosuppressed, chronic opioid users, or elderly patients). A plain abdominal film might be used initially to localize the level of obstruction (proximal vs. distal) and also as baseline for patients' undergoing nonoperative management. It is the authors' preference to initially evaluate patients with SBO with plain abdominal films unless there is a clear indication for emergency exploration. Airfluid levels and bowel loops in the same place on supine and upright films indicate fixed adhesions.

Computed tomography (CT) scan with IV contrast is an important tool in diagnosis and preoperative planning in stable patients without indication for emergency exploration. CT may delineate a lesion, level of obstruction, and the severity of obstruction including any transition point (Fig. 26.1) or closed loop obstruction, and other radiographic



Fig. 26.1 CT—transition point in setting of complete mechanical small bowel obstruction

evidence of ischemia. CT should be taken with intravenous contrast while enteral contrast is not necessary. In high-grade SBO, oral contrast may delay diagnosis and cause patients discomfort and possibly even complications such as aspiration.

The most common cause of bowel obstruction is postoperative adhesions as previously mentioned. The diagnosis of adhesions is made by locating the point of obstruction (transition between dilated and decompressed bowel) without identifiable cause of obstruction, such as neoplasm, hernia, or inflammatory condition. The ability of CT scan to identify a transition zone ranges from 63–93% [12, 13].

CT imaging features in small bowel obstruction indicating the need for surgical intervention include small bowel ischemia, closed loop obstruction, and perforation. Contrastenhanced CT has a sensitivity of 75-100% and specificity of 61–93% in the identification of ischemic SBO [12, 13]. Radiographic signs of ischemia in bowel obstruction include decreased enhancement, mucosal thumbprinting, bowel wall thickening, mesenteric edema, and pneumatosis intestinalis. Closed loop obstruction results from obstruction of a segment of bowel at two different points, and, as a consequence, this obstructed segment of small bowel is isolated from remainder of the gastrointestinal tract. CT scan might differentiate between complete and partial bowel obstruction. Complete SBO is characterized by an abrupt cutoff with air-fluid levels that suggests complete obstruction, while the presence of gas throughout the colon suggests ileus versus partial obstruction. CT is also particularly important for those patients with past medical history of cancer, to assess potential recurrence and/or progression and allow proper cancer staging in the setting of SBO.

Although CT is considered a first-line test in diagnosis, it exposes patients to radiation and it is an expensive modality. Abdominal ultrasound (US) has been investigated as a possible option for diagnosing SBO. The advantages of US include rapid diagnosis and serial assessments without radiation exposure. However, it is operator-dependent. Comprehensive and bedside US have sensitivity ranging between 83% and 97.7%, and specificities between 84% and 100% [14]. US has potential limitations such as if the presence of air limits visualization of underlying structures in the presence of dilated loops of bowel. Further testing then becomes necessary when US fails to identify the cause of the SBO. Diagnostic US findings for SBO include distended and collapsed bowel segments in close proximity, free peritoneal fluid, inspissated intestinal content, paradoxical peristalsis, highly reflective fluid within the bowel lumen, bowel wall edema between serosa and mucosa, or a fixed aperistaltic loop concerning ischemia.

Treatment and Operative Intervention

All patients should have initial management with volume resuscitation, bowel decompression, and rest with correction of metabolic abnormalities (Fig. 26.2). Unless there is indication for immediate intervention, initial pharmacologic management should be centered on antiemetics, analgesics, and antisecretory medications. Prospective trials have evaluated the use of somatostatin with decreased distention and nausea, allowing for effective nasogastric tube decompression and symptom management in poor surgical candidates [15, 16]. After the initial conservative management, different treatments options are available depending on the etiology of SBO.

Adhesive Disease Current guidelines for the management of adhesive SBO recommend that in the absence of signs of bowel ischemia, non-operative management should be tried for up to 3 days as long as the patient does not deteriorate with symptoms suggesting intestinal ischemia [17, 18]. Nonoperative management consists on bowel rest and decompression using nasogastric tube or long intestinal tube, analgesics, and antiemetics if necessary. Conservative management is effective in approximately 70–90% cases [16].

Possible adjunct to nonoperative management is the use of water-soluble contrast agent (WSCA). WSCA creates an osmotic gradient, which then increases bowel transit. A recent systematic review and meta-analysis [19] has shown that WSCA has a sensitivity of 92% and a specificity of 93% in predicting resolution of bowel surgery and that diagnostic accuracy increased significantly if abdominal X-ray is taken after 8 h from the administration. The administration of WSCA reduces the need for surgery (OR 0.55, p = 0.003), length of stay (-2.18 days, p < 0.00001), and time to resolution (-28.25 h, p < 0.00001), without adding morbidity and mortality.

If conservative approach fails after 3 days and/or patients deteriorates, the next step is surgical management, which includes adhesiolysis and small bowel resection if necessary. Although laparoscopic approach might be considered as first option, especially in patients with previous minimally invasive approach, the feasibility of a laparoscopic procedure in the setting of an abdomen with distended loops of bowel and potential complex adhesions could increase the risk for patients' complications. Some authors have reported an increased risk of bowel injury and need for bowel resection in laparoscopic adhesiolysis compared to open approach (53.3% lap vs. 43.3% open). Predictors for a successful laparoscopic adhesiolysis are the following: <2 previous laparotomies, appendectomy as past operation, no previous median laparotomy and single adhesive band [20]. Laparoscopic approach should be carefully evaluated in patients with previous oncologic abdominal procedure, where extensive procedure might have created adhesions too extensive to be treated with a minimally invasive approach.

Small Bowel Intussusception In patients with a small bowel intussusception indications to proceed immediately to the operating room include clinical or radiographic evidence of bowel compromise and peritonitis. In patients with no prior history of operation, cancer should be suspected and a low threshold for prompt surgical exploration is indicated. It is a rare cause of SBO in adults (1–5%) but is secondary to a malignant disease in up to 50% of patients [21] (Figs. 26.3 and 26.4). Strangulation is associated with high mortality rates and conventional signs of vascular compromise may not always be present [22]. Bowel telescoping will then lead to venous and lymphatic congestion, edema, and potentially ischemia and perforation.

Intussusception in the adult necessitates surgical intervention with resection of the involved bowel segment and identification of lead point after thorough evaluation of bowel and peritoneal cavity. Based on fact that half of adult patients with intussusception were noted to have a malignant neoplasm [21], appropriate oncologic workup should be completed.

Intraoperatively, bowel viability may be determined by observation of peristalsis and color, Doppler ultrasound, and occasionally IV fluorescein and Wood lamp. Tumor debulking or even an oncologic resection may be indicated. Operative decision will include whether or not to restore bowel continuity, fashion a stoma, or leave the operating room and come back for a second look laparotomy in the unstable patient. If the lesion is felt to be a primary and resectable neoplasm in a stable patient, the tumor should be resected with a wide margin proximal and distal to the lesion of normal bowel including the lymph node basin for the involved segment.







Fig. 26.3 Intraoperative intussusception with bowel telescoping



Fig. 26.4 Intraoperative intussusception with malignant lead point

Malignant Small Bowel Obstruction (MaSBO) MaSBO is usually caused by locally advanced or metastatic cancer and management is still controversial. Song et al. [9] looked at patients with MasBO from the National Surgical Quality Improvement Program between 2005 and 2017 and compared them to patients with SBO from nonmalignant causes. Of 46,706 patients studied, 1612 (3.5%) had MaSBO. The oncologic patients were more likely to have recent weight loss (22% vs. 4.0%, p < 0.001), severe hypoalbuminemia (18.6% vs. 5.2%, p < 0.001) and pancytopenia (with anemia defined as hematocrit <30%, thrombocytopenia defined as <150 × 10⁹ cells/L and white blood cell count <4 × 109 cells/L). The risk of mortality for the oncologic patients was higher (OR 3.3, p < 0.001). Factors associated with 30-day postoperative mortality are advanced age (over 70), wound status, impaired functional status, low body mass index, low album level (<2.5 g/dL) as a surrogate of nutritional status, smoking history, thrombocytopenia (<150 × 10⁹ cells/L) and elevated WBC count (≥11 × 10⁹ cells).

Considering the high morbidity and mortality, small bowel obstruction in the setting of suspected or known malignancy should include consideration and discussion of goals of care and quality of life in view of life expectancy and overall disease burden. Efforts should be made early in the hospital course to work collaboratively for a treatment plan incorporating patient and family values with a clear discussion regarding the limitations of surgery. A comprehensive and multidisciplinary treatment plan should include consultation with medical oncologists, palliative care specialists, gastroenterologists, radiologists, and dieticians. Considering all factors, up to 50% of patients with MaSBO are deemed inoperable [10]. Surgery might increase overall survival, but morbidity (7-44%), mortality (6-32%), readmission (38-74%) and recurrent obstruction (6-47%) are frequent [23]. In case of surgery, bowel resection when feasible provides the best outcome. Otherwise, bypass should also be considered as well as fecal diversion and gastrostomy tubes for decompression [10].

A palliative, nonoperative approach should be offered for patients with poor performance status, multiple sites of obstruction, carcinomatosis, or ascites. Further details about palliative management of malignant small bowel obstruction can be found in the chapter dedicated to palliative treatment.

Large Bowel Obstruction Large bowel obstruction (LBO), while less common overall than SBO, is more likely to be secondary to malignancy. While SBO can be managed conservatively in approximately 70% cases, LBO requires surgical intervention in 70% of cases. In Western countries, colorectal cancer represents the most common cause of mechanical obstruction and is responsible for 50% of LBO. LBO occurs at initial presentation in 10% of patients with colorectal cancer [24]. The second most common cause for LBO is diverticular strictures accounting for 10–20% of cases. Colonic volvulus, generally cecal or sigmoid, is the third most common cause and it is responsible for 10–17% of cases. Among sites for colonic obstruction, the sigmoid colon is the most common followed by descending colon, splenic flexure, transverse colon, rectum, and ascending colon.

Clinical Presentation and Initial Assessment

Patients with LBO may present similarly to patients with SBO with abdominal distention, absence of flatus and/or bowel movements, and crampy abdominal pain. Physical examination should again pay special attention to evidence of dehydration and hypovolemia as well as hernias with a rectal exam performed to evaluate for the presence of stool in the vault, an obstructing rectal mass or bleeding.

Diagnostic Evaluation

Large bowel obstruction should be worked up similar to SBO. Plain abdominal X-ray has 84% sensitivity and 72% specificity and might be able to reveal the degree of bowel distension and sometimes it is able to localize the presence of transition point. CT abdomen-pelvis with venous contrast is the modality of choice to evaluate the extent and level of obstruction (Fig. 26.5). CT is estimated to have 95% sensitivity and 93% specificity for the diagnosis of LBO.

Treatment and Operative Intervention

All patients should be placed on bowel rest with fluid resuscitation. If the ileocecal valve is incompetent and there is also small bowel distention, nasogastric decompression may be indicated. Partial colonic obstruction should be attempted to be treated conservatively with the hope that a patient can



Fig. 26.5 CT—right colon/cecal dilatation in setting of partial large bowel obstruction

undergo a single-staged planned oncologic resection in an elective situation.

For primary right- sided obstructive colon cancer, general consensus is for primary resection with ileocolonic anastomosis [25, 26]. For unresectable right-sided colon cancer, a side-to-side anastomosis between terminal ileum and transverse colon as an internal bypass can be performed. Alternatively, in the presence of small bowel distension, a loop ileostomy for palliation should be considered. Optimal management of left-sided malignant bowel obstruction is controversial and it is dictated by patient and disease-related factors and available resources and technical skills. Recommendation and guidelines in the literature are at best conflicting [27]. Different surgical options have been explored for left-sided malignant bowel obstruction including primary resection, with or without anastomosis, subtotal colectomy, with or without anastomosis, or diverting stoma. Emergency surgery is associated with mortality rates three times higher than elective resection [28]. Primary resection with/without anastomosis is the most frequent recommended procedure [27]. A Hartmann's resection is an option when the anastomosis is associated with prohibitive risks.

An additional option to bridge to definitive surgery is the use of self-expanding metallic stents (Fig. 26.6) and therefore a gastroenterology consultation may be indicated in the ED. Stent placement can be followed by elective colonic resection (Fig. 26.7) after medical optimization, bowel preparation cancer staging [29, 30]. Self-expanding colonic stents as a bridge to surgery have resulted in high rates of primary anastomosis, less permanent stomas, and lower wound infection rates, without increasing the rate of mortality compared to emergency surgery. In one study, nearly three quarters of patients in the stented left-sided large bowel obstruction group underwent successful one-stage operation versus closer to one quarter with emergency surgery. Initial technical success rates have been quoted as high as 90% with prompt colonic decompression and overall 30-day mortality of less than 2% [31]. Another important advantage of stenting is that it allows a complete cancer staging and discussion at the multidisciplinary meeting, with consensus about the most appropriate treatment and a definitive procedure performed by the appropriate surgeon. Despite all these theoretical advantages, a recently published phase III randomized controlled trial comparing stenting as a bridge to surgery with emergency surgery for left-sided malignant bowel obstruction did not show a difference in mortality at 1 year [32].

Palliative Treatment At present, majority of guidelines recommend stenting as treatment of choice in case of need for palliative management of left-sided malignant bowel obstruction. In the palliative setting, colonic stenting results in faster resumption of oral intake and shorter hospital stay.



Fig. 26.6 CT-stent deployed in malignant bowel obstruction

Fig. 26.7 Large bowel specimen with stent in place



Complication rates are relatively low and include stent migration, perforation, and re-obstruction [27, 31].

Main reason to prefer stent is that as previously mentioned, emergent colonic surgery is associated with a significantly higher morbidity and mortality than elective surgery. The bowel is often friable and distended with significant stool burden, and the patients are often malnourished. Postoperative complications, re-obstruction rates, and readmission rates are high for palliative malignant bowel obstruction surgery [27]. Should the decision be made to proceed with operative intervention without colonic stenting, patients should be selected based on favorable prognostic features.

Gastric Outlet Obstruction

Advanced malignant disease in the upper gastrointestinal tract might result in gastric outlet obstruction (GOO). The most common malignancies that result in GOO are pancreatic cancer, which accounts for 15–20% cases, periampullary cancer, advanced gastric cancer, duodenal/jejunal cancer, and lymphoma. Less commonly, GOO is seen in cases of gallstone impaction, gastric polyp prolapse, PEG tube migration, or gastric volvulus.

Clinical Presentation and Initial Assessment

Patients will present with progressive symptoms, predominantly nausea and vomiting, often projectile, weight loss, abdominal discomfort, cachexia, poor nutritional status, and severe dehydration. Early symptoms may include early satiety and bloating. Patients with malignant disease may have a fairly short duration of symptoms, and symptoms may be intermittent until obstruction is complete.

Median survival in the palliative setting is usually only 7–20 weeks, and for this reason, it is important to reestablish oral intake quickly to improve quality of life.

Diagnostic Evaluation

Initial evaluation should begin with an abdominal X-ray film. Patients with gastric outlet obstruction will have a gastric bubble if no nasogastric tube has been placed, with little or no air in the small bowel or the colon on radiographic imaging. CT imaging offers the advantage of identifying the obstruction and underlying lesion(s).

Barium swallow is a dynamic study and may distinguish GOO from delayed gastric emptying, anastomotic leak, or gastric fistula in the postoperative setting. Evaluation will generally proceed to EGD for both diagnosis and potential treatment with balloon dilation or self-expanding metallic stenting to improve gastric emptying. Differential diagnosis should include gastric dysmotility as well as dysphagia secondary to chemoradiation and anastomotic leak or gastric fistula in the postoperative setting.

Treatment and Operative Intervention

Therapy for GOO should begin with nasogastric decompression and fluid resuscitation and correction of the metabolic sequelae are often seen. Surgical and GI consultation should be sought for treatment planning.

Surgical gastrojejunostomy, often with a concomitant biliary bypass, is currently performed by a laparoscopic or open approach. Patients who are not considered surgical candidates are referred for endoscopic stent placement. For this reason, majority of the studies that have compared surgical jejunostomy and endoscopic stenting have been biased based on patients' selection. Self-expanding stent is inserted under combined endoscopic and fluoroscopic guidance to restore the patency of the stomach and/or duodenum and allows patients to receive nutrition and hydration via the oral route. Stents have the advantages of a short hospital stay compared to palliative surgery but they are also associated with complications such as bleeding, migration, obstruction, and fractures [33]. Palliative resection results in better long-term outcomes in appropriate candidates. Stenting it is also recommended in patients with GOO secondary to newly diagnosed gastric cancer as a temporizing measure prior to complete oncologic workup and treatment. In patients with GOO secondary to lymphoma, chemotherapy is the first-line indicated therapy. Symptom recurrence with duodenal stents is seen in the majority of patients who survive longer than 6–12 months, and the stent often needs to be replaced.

EUS-guided gastroenterostomy consists on placing a lumen-apposing metal stent under endoscopic ultrasound guidance. According to a recent meta-analysis, EUS-guided gastroenterostomy is an effective and safe minimally invasive treatment for benign and malignant GOO [34]. In contrast with endoluminal stents, EUS-guided gastroenterostomy involves placement of a fully covered stent during a bypass, and the stent is placed away from the tumor, without the risk of tumor ingrowth and occlusion.

Gastrointestinal Bleeding

Acute gastrointestinal bleed (GIB) (Fig. 26.8) may present in the patient with an established cancer diagnosis in a variety of clinical settings. Also, GIB can be the first symptom which allows the diagnoses of cancer such as colorectal cancer.

Bleeding can also be exacerbated or caused by medications such as bevacizumab, nonsteroidal anti-inflammatory drugs (NSAIDs), and anticoagulants, commonly used medications in cancer patients. Anti-inflammatories are often used to treat pain for patients with advanced cancer, but their antiplatelet and anticoagulant properties should be considered. Moreover, patients with advanced cancer are often on anticoagulants such as warfarin or enoxaparin, and the considerations of the risks of further bleeding against the risks of deep venous or pulmonary thromboembolism should be considered. Among patients on anticoagulation, patients with cancer develop bleeding complications at a higher rate than those without cancer.

The effect of chemotherapy agents and radiation therapy on thrombocytopenia should also be considered, as this may increase the risk of bleeding. If considered a critical contributor, these agents may be held to allow bone marrow recovery and resolution of thrombocytopenia.

Bleeding is categorized as upper gastrointestinal bleeding (UGIB) proximal to the ligament of Treitz [35, 36] versus lower gastrointestinal bleeding (LGIB) [37]. UGIB is associated with significant morbidity and differential includes neoplasms including GISTs, esophageal and gastric varices, Mallory-Weiss tears, acute hemorrhagic gastritis and gastric/duodenal ulcers. Diagnostic differential for LGIB includes diverticular disease, neoplasms, radiation proctitis, inflammatory bowel disease (IBD), ischemia, infectious colitis, anorectal disease, coagulopathy, and arteriovenous malformations.

Fig. 26.8 Assessment and management of gastrointestinal bleeding algorithm



Clinical Presentation and Initial Assessment

Patients with acute UGIB may present with melena or hematemesis. Patients may endorse symptoms of hypovolemia including dizziness, dyspnea, or chest pain. Hematochezia might occur as well in patients with coagulation disorders. A history of prior episodes of bleeding is significant as 60% of patients may bleed again from the same source. Special attention should be paid to use of medications that may interfere with coagulation or alter hemodynamic response like beta-blockers. Malignancy should be suspected in a patient with a history of smoking, alcohol abuse, or *H. pylori* infection. Physical exam should begin with vital signs and an evaluation of hypovolemia. Evidence of jaundice, caput medusae, or ascites may point to hepatic disease.

Patient with acute LGIB presents usually with hematochezia, which can be mixed with clots as well. A history of previous episodes of rectal bleeding is common, or recent proctological interventions such as hemorrhoidectomy. LGIB might be present also in patients with previous pelvic malignancies treated with radiotherapy or radiochemotherapy. Radiation might cause radiation proctitis, which can be distinguished in two forms, the acute and chronic. Acute radiation proctitis occurs within 3 months of radiation therapy in 13% of patients, it is usually self-limiting and characterized by diarrhea, urgency, tenesmus, and rectal bleeding. Chronic radiation proctitis occurs in 5–10% of patients after 3 months of therapy completion and the typical complaint is rectal bleeding, while less frequent might cause rectal stenosis. Ulcerative or Crohn's colitis can present with bleeding and are a risk factor for the development of colorectal cancer. In case of rectal bleeding, a complete abdominal exam including rectal exam with proctoscopy should be performed. Even in this case, attention should be paid to current patient's medications.

Diagnostic Evaluation

Initial ED evaluation should include physiological parameters, such as heart rate, blood pressure, respiratory rate, oxygen saturation, and mental status. Initial management should ensure adequate IV access, either large-bore peripheral IV lines or a central line as indicated. Resuscitation is essential and coagulopathies should be corrected. Urinary output and mental status, markers of end-organ perfusion, should be monitored. Early GI consultation is recommended.

A nasogastric tube is often placed for gastric lavage. A positive aspirate of gross blood or coffee-ground appearance, indicated UGIB should be followed by esophagogastroduodenoscopy (EGD). A bilious aspirate should be followed by colonoscopy. Blood tests should include hemoglobin level, hematocrit, platelet count, prothrombin time, INR, BUN, creatinine, electrolytes, liver function tests, and blood cross-matching.

Treatment and Operative Intervention

PPI therapy should be initiated at the suspicion for UGIB. Crystalloids are recommended for volume replacement.

Coagulopathies should be corrected in these cases. Severe thrombocytopenia may lead to spontaneous GI hemorrhage. The American Society of Clinical Oncologic recommends a prophylactic platelet threshold for transfusion: $10,000/\mu$ L for adult patients with leukemia, and multiple studies have found decreased frequency and severity in gastrointestinal hemorrhage in patients with platelet counts above $20,000/\mu$ L versus $10,000/\mu$ L or $5000/\mu$ L [38].

EGD and colonoscopy may be both therapeutic and diagnostic. Erythromycin can be given prior to endoscopy to facilitate gastric emptying, decreasing the need for multiple endoscopic evaluations. Therapeutic endoscopic maneuvers include injection of vasoconstrictors or sclerotherapy, thermal coagulation, and mechanical occlusion of bleeding sites (including clips, bear claw, and over-the-scope clips application). In cancer patients with cirrhotic liver disease, rubber banding, sclerotherapy, or temporizing balloon tamponade may control hemorrhage.

Early endoscopy (<8–12 h from presentation) is ideal in a stable patient and will offer the best chance of localizing and identifying the source of GI bleed and intervening. Therapeutic intervention has been demonstrated to be most successful when performed within 12 h, with declining results as time passed. If bleeding recurs despite medical and endoscopic therapy, endoscopic intervention should be repeated.

Other options for localizing GIB include CT angiography (CTA) which may provide helpful information, including hyperdensity of the mesenteric fat, contrast enhancement of the bowel wall, vascular extravasation of contrast, thickening of the bowel wall, polyps, tumors, and vascular ectasia. CTA has been noted to detect arterial bleeding at rates as low as 0.5 mL/min.

Mesenteric angiography (for bleeding of at least 1 mL/ min) may be useful for poor operative candidates where other measures have failed. Should embolization also fail during mesenteric angiography, and operative intervention ultimately required, the area may be still be identified for operative intervention with methylene blue infusion. A recent meta-analysis evaluating embolization therapy versus surgery for non-variceal UGIB found a lower complication rate for embolization (31% vs. 50%) but a higher rate of rebleeding (35% vs. 18%) [39]. Arteriography may be therapeutic but requires active bleeding of more than 1 mL/min and should be reserved for patients with massive, ongoing bleeding in whom endoscopy is not feasible or colonoscopy fails to reveal the source of the hemorrhage.

Colitis may also be a special consideration in the patient undergoing radiation or chemotherapy. Radiation enteritis may lead to LGIB with a minority of patients even requiring hospitalization. Most patients with colonic ischemia respond to bowel rest, IV fluids, and antibiotics.

Rectal bleeding in patients with radiation proctitis might require endoscopic and/or surgical treatment when conservative measures have failed [40]. The aim of endoscopy is to obliterate telangiectasias using different methods, such as laser and argon plasma coagulation, radiofrequency ablation, or cryotherapy. Endoscopy is successful in up to 90% of patients and it is considered the most effective treatment for bleeding associated with radiation proctitis. Surgery is the last resort and can include fecal diversion via colostomy or ileostomy to rescue bleeding and improve other associated symptoms such as tenesmus, incontinence, and pain. More aggressive procedures like proctectomy have a high complications rate (morbidity 15–80% and mortality 3–9%) and they are rarely performed [40].

Indications for emergent operative intervention include hemodynamic instability despite maximal support measures, substantial bleeding (six units or more), or bleeding that is not controlled endoscopically. Rarely, operative intervention is required. Surgical options for LGIB include segmental colectomy when the source of bleeding can be localized. If no source is found, but bleeding is clinically significant and ongoing, a subtotal colectomy should be performed with an end ileostomy and rectal pouch leaving the patient in discontinuity. Such a situation is associated with high morbidity and mortality in the unstable patient.

Neutropenic Enterocolitis

Patients with cancer often develop neutropenia secondary to chemotherapy treatment. Neutropenia is defined as an absolute neutrophilic count (ANC) <1000/mm³. In patients with hematologic malignancies, neutropenia is commonly due to

failure of bone marrow and therapies damaging the marrow, while in patients with solid tumors, chemotherapy may lead to underproduction of hematopoietic cell lines. Neutropenic patients often have mucositis resulting in reduced enteric mucosal barrier function allowing for bacterial translocation culminating in a characteristic pathology entity known as neutropenic enterocolitis (NEC) or typhlitis.

According to a recent retrospective review [41] conducted over 8 years at MD Anderson Cancer Center among 49,244 patients with neutropenia, 2.7% of them developed NEC, of those 94 (70%) had hematologic malignancies and 48 (36%) underwent hematopoietic stem cell transplantation. Neutropenic enterocolitis can be fatal, resulting in sepsis (11%), focal bowel ischemia and necrosis (2%) and perforation (2%) in patients who are severely immunosuppressed. The cecum is the most commonly affected site likely secondary to its vascularization and distensibility. NEC is associated with severe morbidity and mortality between 74% and 97% [41, 42]. Factors associated with worse prognosis are age, severe neutropenia (<500 cells/µL), prolonged neutropenia, and concomitant systemic infection.

Clinical Presentation and Initial Assessment

The classic presentation of neutropenic enterocolitis is a patient with absolute neutrophil count <1000 cells/ μ L, new onset abdominal pain and fever. Most commonly, it occurs 2–3 weeks after receiving cytotoxic chemotherapy, when neutropenia is most profound. Patients may have nausea, vomiting, diarrhea, and hematochezia. Patients with prior episodes of neutropenic enterocolitis are at risk for recurrent episodes.

Differential diagnosis should include graft-versus-host disease after allogeneic hematopoietic cell transplantation, infectious colitis including *Cytomegalovirus* and *C. difficile* colitis and *Norovirus* in immunocompromised hosts, ischemic colitis (more commonly left-sided), appendicitis and colonic pseudoobstruction.

Diagnostic Workup

Evaluation should begin with full infectious workup including blood and stool cultures and *C. difficile* toxin assays. Endoscopic findings include ulcerative inflammation, which is present in 50% of cases, nonulcerative inflammation and active bleeding in 40% of cases; colonoscopy is relatively contraindicated and may cause cecal perforation. CT findings may include bowel wall thickening, mesenteric stranding, bowel dilatation, mucosal enhancement, and pneumatosis (Fig. 26.9).



Fig. 26.9 CT—small bowel dilation with diffuse wall thickening and mucosal enhancement in setting of neutropenic enterocolitis

Treatment and Operative Intervention

First-line treatment is conservative, with nasogastric decompression when significant small bowel dilatation is present, intravenous fluids, nutrition support, broad-spectrum antibiotics and blood product support as needed. Broad-spectrum antibiotic regimen should include agents that are active against *Pseudomonas aeruginosa, Escherichia coli*, other enteric Gram-negative bacilli and anaerobes. Coverage for *C. difficile* should be included until ruled out. Antifungal coverage should be considered. Antidiarrheal and narcotic pain medications should be avoided. Sepsis is the most common cause for mortality in this group of patients.

Surgical intervention is indicated for patients with bowel ischemia, free perforation, peritonitis, and severe hemorrhage despite correction of coagulopathies. The most common operation performed is right hemicolectomy with formation of stoma, while extended colectomy is rarely indicated and should be avoided.

Radiation Enteritis

Radiation enteritis is an inflammatory process within the intestinal mucosa as a consequence of radiation exposure and might affect both small and large bowel injury due to radiotherapy leading to inflammation, edema, and decreased bowel function. Radiation enteritis and radiation proctitis may be classified as acute versus chronic and localized versus diffuse. Despite attempts to protect the bowel from the radiation field when possible, gastrointestinal epithelium is especially susceptible to injury given its high proliferative rate. Cell damage in the mucosa leads to microvascular damage, inflammation, edema, and decreased absorptive capac-
ity. Initial damage may be seen in hours and continue for weeks. Damage to the intestinal mucosa may lead to fibrosis, perforation, fistulae, or abscess.

There is a lack of consensus about diagnostic criteria and often symptoms are underreported by patients; as a consequence, only a minority of them are reviewed by gastroenterologists. Despite lack of diagnosis, 90% of patients have gastrointestinal symptoms in the first few weeks after receiving abdomino-pelvic radiotherapy and almost 90% of patients report chronic changes in their bowel habit with important consequences in quality of life due to these symptoms. Acute radiation enteritis may occur even transiently in up to 75% of patients undergoing radiation therapy for abdominal and pelvic cancers. Chronic radiation enteritis may occur in closer to 5–20% of patients.

Risk factors might be divided in treatment-related, including radiation dose and fractionation schedule, treatment field size, and intestinal volume irradiated, which is the main determinant of intestinal radiation-induced toxicity. Other patient-related risk factors are: diminished splanchnic perfusion secondary to diabetes mellitus or atherosclerotic disease, hypertension, tobacco smoking, BMI lower than 30, and advanced age. Concomitant administration of chemotherapy is another factor due to increased toxicity for cumulative effect. Previous abdominal surgery is an increased risk of radiation enteritis due to fixed bowel position due to postoperative adhesions.

Clinical Presentation and Initial Assessment

Acute radiation enteritis usually occurs in the second week post-radiotherapy, with a peak on week 4–5. Clinical manifestations include nausea, vomiting, abdominal pain, diarrhea associated with blood and mucus, hematochezia, tenesmus, and incontinence. A possible consequence is systemic sepsis due to bacterial translocation through inflamed intestinal mucosa. Given the diminished absorption compounding an often preexisting cancer-related anorexia, patients often are dealing with malnutrition and weight loss. Patients may bleed from ulceration, have signs of systemic infection from abscess, or present with obstructive symptoms.

Chronic enteritis might occur between 2 months and several years after radiation treatment. Symptoms include chronic abdominal pain, malabsorption, and weight loss, diarrhea, GI bleeding and complications secondary to strictures and fistulae such as obstruction and perforation. Differential diagnosis for the chronic form includes tumor recurrence, inflammatory bowel disease (ulcerative colitis, Crohn's disease), ischemic colitis, infectious colitis, STD proctitis (e.g., lymphogranuloma venereum, and gonorrhea) and inflammatory bowel syndrome.

Diagnostic Evaluation

The diagnosis of acute form of radiation enteritis is mainly clinical, with symptoms previously described, which might be only transient. If more severe a CT scan should be performed to rule out possible serious consequences, such as perforation and or obstruction.

The chronic form might be diagnosed again through patients' symptoms. Obtaining an abdominal CT/MRI is important because both sensitive at delineating subtle strictures and mucosal irregularities as well as cancer recurrence. A further endoscopy evaluation might be considered to evaluate mucosal inflammation, ulceration and possible fistulae/ stricture formation, while allowing biopsies.

Treatment and Operative Intervention

Most cases of acute radiation enteritis are self-limited and should be treated with supportive therapy. Initial management includes elemental diet, IV hydration, bowel rest and \pm octreotide. Diarrhea can be reduced by the use of bulking agents and antimotility drugs such as loperamide. Antispasmodics and antiemetics might be used for symptoms relief; opioids have the dual function of reducing intestinal motility and providing analgesia. Persistent symptoms despite medical interventions may require a surgical consultation.

In patients with chronic radiation enteritis, conservative management as above is employed, however, up to 30% require surgical intervention to relieve symptoms of obstruction secondary to luminal strictures and to manage fistulae, perforation and/or bleeding. As this is a high-risk intervention with postoperative morbidity and mortality rates of 30% and 5% respectively [43], patient selection is important.

Conclusion

The acute abdomen in the patient dealing with either a new or established cancer diagnosis creates a special challenge for the ED physician to individualize decisions, keeping in mind goals of care and the patient's wishes. One must maintain a broad differential diagnosis and cancer-specific considerations need to be maintained. Evaluation and intervention should proceed with the involvement of a multidisciplinary care team with the patient in the center. Caring for a cancer patient in the setting of an acute change in condition ultimately means knowing not only what is possible surgically and what is indicated medically, but also evaluating what is right for the patient overall.

References

- Ray NF, Denton WG, Thamer M, Henderson SC, Perry S. Abdominal adhesiolysis: inpatient care and expenditures in the United States in 1994. J Am Coll Surg. 1998;186:1–9.
- Scott JW, Olufajo OA, Brat GA, Rose JA, Zogg CK, Haider AH, et al. Use of national burden to define operative emergency general surgery. JAMA Surg. 2016;151(6):e160480.
- ESCP Intestinal Failure Group, Vaizey CJ, Maeda Y, Barbosa E, Bozzetti F, Calvo J, Irtun Ø, et al. European Society of coloproctology consensus on the surgical management of intestinal failure in adults. Color Dis. 2016;18(6):535–48.
- Lee MJ, Sayers AE, Drake TM, Marriott PJ, Anderson ID, Bach SP, NASBO Steering Group and NASBO Collaborators, et al. National prospective cohort study of the burden of acute small bowel obstruction. BJS Open. 2019;3(3):354–66.
- National Audit of Small Bowel Obstruction Steering Group & National Audit of Small Bowel Obstruction Collaborators. Outcomes following small bowel obstruction due to malignancy in the national audit of small bowel obstruction. Eur J Surg Oncol. 2019;45(12):2319–24.
- Anthony T, Baron T, Mercadante S, Green S, Chi D, Cunningham J, et al. Report of the clinical protocol committee: development of randomized trials for malignant bowel obstruction. J Pain Symptom Manag. 2007;34(1 Suppl):S49–59.
- Cousins SE, Tempest E, Feuer DJ. Surgery for the resolution of symptoms in malignant bowel obstruction in advanced gynaecological and gastrointestinal cancer. Cochrane Database Syst Rev. 2016;(1):CD002764.
- Prost À la Denise J, Douard R, Malamut G, Mecheri F, Wind P. Small bowel obstruction in patients with a prior history of cancer: predictive findings of malignant origins. World J Surg. 2014;38(2):363–9.
- Song Y, Metzger DA, Bruce AN, Krouse RS, Roses RE, Fraker DL, et al. Surgical outcomes in patients with malignant small bowel obstruction: a national cohort study. Ann Surg. 2020; https://doi. org/10.1097/SLA.000000000003890. Epub ahead of print.
- Krouse RS. Malignant bowel obstruction. J Surg Oncol. 2019;120(1):74–7.
- Roses RE, Folkert IW, Krouse RS. Malignant bowel obstruction: reappraising the value of surgery. Surg Oncol Clin N Am. 2018;27(4):705–15.
- Taylor MR, Lalani N. Adult small bowel obstruction. Acad Emerg Med. 2013;20(6):528–44.
- Colon MJ, Telem DA, Wong D, Divino CM. The relevance of transition zones on computed tomography in the management of small bowel obstruction. Surgery. 2010;147(3):373–7.
- Gottlieb M, Peksa GD, Pandurangadu AV, Nakitende D, Takhar S, Seethala RR. Utilization of ultrasound for the evaluation of small bowel obstruction: a systematic review and meta-analysis. Am J Emerg Med. 2018;36:234–42.
- Khoo D, Hall E, Motson R, Riley J, Denman K, Waxman J. Palliation of malignant intestinal obstruction using octreotide. Eur J Cancer. 1994;30A(1):28–30.
- Mangili G, Franchi M, Mariani A, Zanaboni F, Rabaiotti E, Frigerio L, et al. Octreotide in the management of bowel obstruction in terminal ovarian cancer. Gynecol Oncol. 1996;61(3):345–8.
- Ten Broek RPG, Krielen P, Di Saverio S, et al. Bologna guidelines for diagnosis and management of adhesive small bowel obstruction (ASBO): 2017 update of the evidence-based guidelines from the world society of emergency surgery ASBO working group. World J Emerg Surg. 2018;13:24.
- Schraufnagel D, Rajaee S, Millham FH. How many sunsets? Timing of surgery in adhesive small bowel obstruction: a study

of the Nationwide Inpatient Sample. J Trauma Acute Care Surg. 2013;74(1):181–7; discussion 187–9.

- Ceresoli M, Coccolini F, Catena F, Montori G, Di Saverio S, Sartelli M, Ansaloni L. Water-soluble contrast agent in adhesive small bowel obstruction: a systematic review and meta-analysis of diagnostic and therapeutic value. Am J Surg. 2016;211(6):1114–25.
- Farinella E, Cirocchi R, La Mura F, Morelli U, Cattorini L, Delmonaco P, et al. Feasibility of laparoscopy for small bowel obstruction. World J Emerg Surg. 2009;4:3.
- Hanan B, Diniz TR, da Luz MM, da Conceição SA, da Silva RG, Lacerda-Filho A. Intussusception in adults: a retrospective study. Color Dis. 2010;12(6):574–8.
- Sarr MG, Bulkley GB, Zuidema GD. Preoperative recognition of intestinal strangulation obstruction. Prospective evaluation of diagnostic capability. Am J Surg. 1983;145:176–82.
- Paul Olson TJ, Pinkerton C, Brasel KJ, Schwarze ML. Palliative surgery for malignant bowel obstruction from carcinomatosis: a systematic review. JAMA Surg. 2014;149(4):383–92.
- Alavi K, Field C. Large bowel obstruction. In: Steel S, Hull T, Read T, Saclarides T, Senagore A, Whitlow C, editors. The ASCRS textbook of colon and rectal surgery. Cham: Springer International; 2016. p. 669–95.
- 25. Finan PJ, Campbell S, Verma R, MacFie J, Gatt M, Parker MC, et al. The management of malignant large bowel obstruction: ACPGBI position statement. Color Dis. 2007;9(Suppl 4):1–17.
- Pisano M, Zorcolo L, Merli C, Cimbanassi S, Poiasina E, Ceresoli M, et al. 2017 WSES guidelines on colon and rectal cancer emergencies: obstruction and perforation. World J Emerg Surg. 2018;13:36.
- Webster PJ, Aldoori J, Burke DA. Optimal management of malignant left-sided large bowel obstruction: do international guidelines agree? World J Emerg Surg. 2019;14:23.
- Morris EJ, Taylor EF, Thomas JD, Quirke P, Finan PJ, Coleman MP, et al. Thirty-day postoperative mortality after colorectal cancer surgery in England. Gut. 2011;60(6):806–13.
- 29. Matsuda A, Yamada T, Matsumoto S, Sakurazawa N, Kawano Y, Sekiguchi K, et al. Short-term outcomes of a self-expandable metallic stent as a bridge to surgery vs. a transanal decompression tube for malignant large-bowel obstruction: a meta-analysis. Surg Today. 2019;49(9):728–37.
- Pirlet IA, Slim K, Kwiatkowski F, Michot F, Millat BL. Emergency preoperative stenting versus surgery for acute left-sided malignant colonic obstruction: a multicenter randomized controlled trial. Surg Endosc. 2011;25(6):1814–21.
- 31. Van den Berg MW, Sloothaak DA, Dijkgraaf MG, van der Zaag ES, Bemelman WA, Tanis PJ, et al. Bridge-to-surgery stent placement versus emergency surgery for acute malignant colonic obstruction. Br J Surg. 2014;101(7):867–73.
- 32. Hill J, Kay C, Morton D. CREST: randomized phase III study of stenting as a bridge to surgery in obstructing colorectal cancer: results of the UK ColoRectal endoscopic stenting trial (CREST) (abstract). J Clin Oncol. 2016;34(15 Suppl):3507. https://ascopubs. org/doi/abs/10.1200/JCO.2016.34.15_suppl.3507.
- Jang S, Stevens T, Lopez R, Bhatt A, Vargo JJ. Superiority of gastrojejunostomy over endoscopic stenting for palliation of malignant gastric outlet obstruction. Clin Gastroenterol Hepatol. 2019;17(7):1295–302.
- 34. McCarty TR, Garg R, Thompson CC, Rustagi T. Efficacy and safety of EUS-guided gastroenterostomy for benign and malignant gastric outlet obstruction: a systematic review and meta-analysis. Endosc Int Open. 2019;7(11):E1474–82.
- Sverdén E, Markar SR, Agreus L, Lagergren J. Acute upper gastrointestinal bleeding. BMJ. 2018;363:k4023.
- Barkun AN, Almadi M, Kuipers EJ, Laine L, Sung J, Tse F, et al. Management of nonvariceal upper gastrointestinal bleeding: guide-

line recommendations from the international consensus group. Ann Intern Med. 2019;171(11):805–22.

- 37. Oakland K, Chadwick G, East JE, Guy R, Humphries A, Jairath V, McPherson S, Metzner M, Morris AJ, Murphy MF, Tham T, Uberoi R, Veitch AM, Wheeler J, Regan C, Hoare J. Diagnosis and management of acute lower gastrointestinal bleeding: guidelines from the British Society of Gastroenterology. Gut. 2019;68(5):776–89.
- Schiffer CA, Anderson KC, Bennett CL. Platelet transfusion for patients with cancer: clinical practice guidelines of the American Society of Clinical Oncology. J Clin Oncol. 2001;19:1519–38.
- 39. Tarasconi A, Baiocchi GL, Pattonieri V, et al. Transcatheter arterial embolization versus surgery for refractory non-variceal upper

gastrointestinal bleeding: a meta-analysis. World J Emerg Surg. 2019;14:3.

- Tabaja L, Sidani SM. Management of radiation proctitis. Dig Dis Sci. 2018;63(9):2180–8.
- Abu-Sbeih H, Ali FS, Chen E, Mallepally N, Luo W, Lu Y, et al. Neutropenic enterocolitis: clinical features and outcomes. Dis Colon Rectum. 2020;63(3):381–8.
- Rodrigues FG, Dasilva G, Wexner SD. Neutropenic enterocolitis. World J Gastroenterol. 2017;23(1):42–7.
- Hale MF. Radiation enteritis: from diagnosis to management. Curr Opin Gastroenterol. 2020;36(3):208–14.



Sai-Ching Jim Yeung



27

Case Study

A 36-year-old woman with stage 2 HER2-positive hormone receptor-negative breast cancer presented to the emergency department (ED) with the complaint of severe diarrhea. She had completed neoadjuvant chemotherapy and underwent a left mastectomy with left axillary lymph node dissection 10 days ago. She had mild postoperative wound erythema and had completed a 7-day course of oral clindamycin 3 days ago. She started having watery diarrhea 3 days ago. Two days ago, she was evaluated by the surgical team in the clinic. A stool sample was sent for *Clostridium difficile* toxin analysis, and she was empirically started on metronidazole 500 mg orally every 8 hours. Over the last 2 days, her diarrhea worsened, and she had abdominal cramping and 10 stools in the past 24 hours. She was feeling weak in general and nauseous. She denied having fever, chills, or vomiting.

In the ED, her vital signs demonstrated mild tachycardia and orthostatic hypotension. She had hyperactive bowel sounds, and her abdominal examination was otherwise benign. The report of the analysis of the stool sample collected 2 days ago showed the presence of *C. difficile* toxin. Her laboratory results were remarkable for a serum potassium of 3.0 mEq/L, BUN 40 mg/dL, and creatinine 1.4 mg/dL. Abdominal radiographs showed a normal nonobstructive gas pattern.

She was admitted for intravenous fluid hydration and potassium replacement. Stool samples were collected for stool culture. Metronidazole was discontinued and she was started on oral vancomycin.

Clostridium difficile diarrhea or colitis is a well-known complication of antibiotic therapy. It is also a very common problem among cancer patients. Metronidazole treatment for *C. difficile*, however, has an approximately 20% rate of lack of clinical response and an approximately 30% rate of recur-

Department of Emergency Medicine, Division of Internal Medicine, The University of Texas MD Anderson Cancer Center, Houston, TX, USA e-mail: syeung@mdanderson.org rence within 3 months. Despite the suboptimal response rate, metronidazole is a popular first-line treatment because it is safe and inexpensive.

Introduction

Diarrhea is the frequent (>3/day) passage of loose stools with urgency and it is a frequent comorbidity or adverse event associated with therapy in cancer patients. Diarrhea in cancer patients can be severe; chronic diarrhea can cause electrolyte abnormalities and malnutrition, while uncontrolled diarrhea may lead to severe dehydration and lifethreatening electrolyte abnormalities. The need to avoid recurrence of serious diarrheal complications may lead to dose reduction or discontinuation of antineoplastic therapies. Cancer patients frequently present to emergency departments with acute or chronic diarrheal complications that require emergency evaluation and treatment.

Causes

Although cancer and cancer treatments (chemotherapy, radiation therapy, surgery, targeted therapy, and immunotherapy) can cause diarrhea, emergency care providers must not forget causes unrelated to cancer, e.g., lactose intolerance, food poisoning, viral gastroenteritis, side effects of noncancer drugs, inflammatory bowel disease, and irritable bowel syndrome (Table 27.1). In this chapter, we focus on causes relevant to cancer and cancer treatments.

Paraneoplastic Syndromes

Certain cancers can cause diarrhea by secreting hormones, including the following:

- · Carcinoid tumors
- Zollinger-Ellison syndrome (gastrinoma)

© Springer Nature Switzerland AG 2021

S.-C. J. Yeung (🖂)

K. H. Todd et al. (eds.), Oncologic Emergency Medicine, https://doi.org/10.1007/978-3-030-67123-5_27

 Table 27.1
 Causes of diarrhea in cancer patients

Paraneoplastic	Carcinoid syndrome, medullary carcinoma of the thyroid, neuroendocrine pancreatic cancer (e.g., gastrinoma, VIPoma), pheochromocytoma
Surgery related	Celiac plexus block, cholecystectomy,
	esophagogastrectomy, gastrectomy,
	pancreaticoduodenectomy (Whipple procedure),
	intestinal resection (malabsorption due to short
	bowel syndrome), vagotomy
Chemotherapy	Bortezomib, capecitabine, 5-fluorouracil,
	oxaliplatin, carboplatin, cisplatin, cytosine
	arabinoside, cyclophosphamide, daunorubicin,
	inipatasen, docetaxel, doxorubicin, methotrexate,
Torgatad	Constraint (25.0, 51.6% [oll grades], 1, 2% [grades]
therapy ^a	3 (1) Gentinib (23.9–31.6% [all grades], 1–3% [grades
ulciapy	(18% 3%)
	Afatinib (95% 14.4%)
	Lapatinib (64%, 10%)
	Trastuzumab (3.7% [grades 3–4])
	Pertuzumab (3% [grades 3–4])
	Imatinib (20–26%, 1%)
	Pazopanib (<4%, 52%)
	Regorafenib (34–40%; 5–8%)
	Cabozantinib (64%, 12%)
	Sunitinib (44%, 5%)
	Sorafenib (55.3%, 7.8%)
	Z_{1V} -anibercept (69.2%, 19%)
	Axiumb $(11\% [grades 5-4])$
	$ F_{\text{variation}} (1\% [\text{grades } 3_4]) $
	Vemurafenih (5–6% [all grades])
	Dabrafenib (1% [all grades])
	Trametinib (45–50%, 4%)
	Selumetinib (45–50%, 4%)
	Crizotinib (60% [all grades])
	Bortezomib (51%, 8%)
Radiation	Radiation therapy to the abdomen, pelvis,
	para-aortic lymph nodes, lumbar spine
Bone marrow	Conditioning chemotherapy, total body irradiation,
transplantation	graft-versus-host disease after allogeneic
	transplants
Immune	Ipilimumab, cemiplimab, pembrolizumab,
checkpoint	nıvolumab, atezolizumab, durvalumab, avelumab
Inhibitors	Number of the state territies and the state D - 11
Infection	Neutropenic contis, typnitus, and neutis. Bactitus
	Clostridium perfringens Cryptosporidium
	<i>Cytomegalovirus</i> (in immunocompromised hosts).
	<i>Giardia lamblia</i> . <i>Rotavirus</i> . <i>Salmonella</i> . <i>Shigella</i>
Fecal impaction	Liquid stool going around impacted stool
Comorbid	Diabetes mellitus, hyperthyroidism.
diseases	gastroenteritis, irritable bowel syndrome, celiac
	disease, inflammatory bowel disease (Crohn's
	disease, ulcerative colitis), HIV/AIDS
Psychological	Stress
factors	

^aPercentage inside parentheses indicate the incidence rate of diarrhea of all grades and severe grades (grades 3–4), respectively

- VIPomas (neuroendocrine tumors that secrete vasoactive intestinal peptide [VIP] autonomously)
- Medullary thyroid carcinoma (sporadic, familial, or as part of Sipple syndrome—multiple endocrine neoplasia type 2 [MEN 2])
- Neuroendocrine tumors as part of the Wermer syndrome—multiple endocrine neoplasia type 1 (MEN 1)

Treatment-Induced Diarrhea

Chemotherapy In addition to cancer cells, cytotoxic chemotherapy kills other fast-growing cells, including those in the intestinal lining. Certain chemotherapeutic agents can disturb the normal absorptive and secretory functions of the small bowel, resulting in treatment-related diarrhea [1]. Chemotherapeutic agents associated with severe diarrhea include fluorouracil, capecitabine, irinotecan, paclitaxel, docetaxel, vinorelbine, etc. The diarrhea caused by irinotecan may be delayed (>24 hours) and severe. Concomitant abdominal or pelvic radiotherapy and recent gastrointestinal surgery are associated with increased severity of treatmentinduced diarrhea. Platinum-based therapy can cause diarrhea and colitis. The median time from starting platinum-based therapy to colitis is 66 days and the median duration is 20 days. Colonoscopic examination will show ulceration in one third of cases and nonulcerative inflammation in the colonic mucosa in another one third of cases. About half of patients with platinum drug-induced colitis will require hospitalization [2]. Taxanes also causes colitis. The median time from start of taxanes to colitis symptom is 31 days. Colonoscopy also shows inflammation, but some patients have microscopic colitis [3].

Targeted Therapy Diarrhea induced by anti-EGFR targeted therapy is secretive and due to excess chloride secretion. For small molecule tyrosine kinase inhibitors (TKIs) targeting EGFR (see Table 27.1), diarrhea of all grades occurs in up to 60% of patients [4] and as many as 10% of cases are severe. Anti-EGFR monoclonal antibodies (mAb) (cetuximab and panitumumab) also causes diarrhea of all grades in almost a quarter of patients, and about 1–2% have severe diarrhea. In non–small cell lung cancer, the combination of anti-EGFR TKI and antiangiogenic therapy results in improved antineoplastic activity; however, these combinations also increase diarrhea [5]. The receptor tyrosine kinase (KIT) is highly expressed in the interstitial cells of Cajal, which are pacemakers for intestinal motility. Some TKIs (e.g., imatinib and sunitinib) may induce diarrhea through inhibition of KIT. Cyclin-dependent kinase (CDK) 4/6 inhibitors (palbociclib, ribociclib, and abemaciclib) cause diarrhea of all grades in about 20% of patients and about 1% are severe [6]. PI3K inhibitors cause a watery nonbloody diarrhea in over 50% of patients [7]. An early onset, less severe diarrhea may occur at a median of 1.9 months after starting PI3K inhibitors, while a more severe diarrhea occurs at a median of 7.1 months after starting treatment. The severe diarrhea is due to immune-mediated colitis [8].

Radiotherapy External beam radiotherapy (XRT) that focuses on the thoracolumbar spine, para-aortic lymph nodes, abdomen and pelvis for cervical cancer, colorectal cancer, prostate cancer, metastatic cancer, etc., as well as pelvic brachytherapy exposes part of the intestine to radiation, causing diarrhea. Diarrhea is the most common side effect of radiotherapy. The overall prevalence of irradiation-induced diarrhea may reach 35% [9]. Factors predicting the severity of XRT-induced diarrhea include total radiation dose, fractionation, the volume of bowel exposed to radiation (related to the method of radiation such as intensity-modulated and image-guided radiotherapy [IMRT] and volumetric modulated arc therapy [VMAT]), and concurrent chemotherapy. Acute diarrhea may occur at about 10 Gy and last up to 3 months after treatment. Chronic radiation enteritis can begin months or even years after treatment. Bile acid malabsorption may be responsible for some portion of radiationinduced diarrhea [10].

Surgery Surgical treatment of cancer may involve removal of sections of the gastrointestinal tract or organs with endocrine and digestive functions. These anatomic changes may limit the ability of the gastrointestinal tract to absorb certain nutrients, e.g., fat, resulting in diarrhea. Bowel resection lessens the surface area for reabsorption of water from food. Pancreatic cancer or its surgical treatment can compromise exocrine pancreatic function, leading to lack of digestion and malabsorption. Surgical changes in biliary anatomy will also compromise emulsification of fat by bile salts and consequently digestion and absorption of fatty food. Diarrhea is also part of the dumping syndrome which occurs when undigested food moves too rapidly into the small intestine. The dumping syndrome is associated with gastrectomy, gastroenterostomy, gastrojejunostomy, vagotomy, pancreaticoduodenectomy, and esophagectomy.

Bone Marrow Stem Cell Transplant In stem cell transplantation, conditioning chemotherapy and total body radiation may cause diarrhea. In autologous peripheral blood stem cell transplantation, in addition to side effects of chemotherapy and radiation, antimicrobials and infectious complications (including C. difficile) can also cause diarrhea. *Clostridium difficile* is the most common pathogen with an incidence of 15% during the 7 days before, and the first 30 days after, autologous transplant [11]. After allogeneic hematopoietic cell transplantation, diarrhea can be a frequent and serious complication. The diagnosis of acute graftversus-host disease as the cause of diarrhea is straightforward if rash, jaundice, nausea, and vomiting are present [12]. However, if only diarrhea is present, the diagnosis may not be so obvious. Nevertheless, infectious (viral, bacterial, and parasitic) causes of diarrhea must be excluded before starting immune-suppressive drugs to treat graft-versus-host disease. Graft-versus-host disease-induced diarrhea usually occurs between 10 and 100 days after transplant and may resolve or become chronic.

Immune Checkpoint Therapy The most frequently seen GI complications secondary to immune checkpoint inhibitor therapy are diarrhea and colitis [13, 14]. Diarrhea can be due to immune-mediated colitis, which mainly involves the descending colon. Although immune checkpoint inhibitor-induced colitis is similar to inflammatory bowel diseases in many aspects, this immune-mediated adverse effect (irAE) can start acutely, progress rapidly, and lead to potential serious complications, including bowel perforation and death [15]. In some cases, enteritis without any colonic involvement may lead to small bowel obstruction [16].

Pancreatic toxicity associated with immune checkpoint inhibitor therapy is uncommon (<2%) and usually presents as a transient asymptomatic increase in lipase or amylase [17, 18]. Acute pancreatitis is rare [17–20] and patients present with a typical pancreatitis picture or with isolated symptoms of nausea, vomiting, fever, epigastric pain, or diarrhea [17]. Pancreatic exocrine insufficiency with or without pancreatitis manifests as irregular stools with diarrhea, discoloration of feces, and weight loss.

Infectious Enteritis Cancer patients are susceptible to infectious diarrhea. Bacteria, viruses, and parasites are all potential culprits. Neutropenia, immune suppressants, antibiotic use, and breakdown of natural defenses against microbes are all factors that increase the risk for cancer patients. *Clostridium difficile* diarrhea or colitis is a common problem.

In neutropenic patients, diarrhea, abdominal pain, and fever are symptoms of neutropenic enterocolitis. The cecum is frequently affected, often extending to the ileum, and occasionally the ascending and transverse colon. Immunosuppression and frequent use of antibiotics in neutropenic patients alters the normal flora, leading to infection by resistant bacteria or unusual bacterial species. Bacteremia is frequent. Gram-negative rods (Escherichia coli, Pseudomonas aeruginosa, Klebsiella, Enterobacteriaceae), gram-positive cocci (Streptococci, Enterococci), fungi (Candida, Aspergillus, or Zygomycetes), and viruses (cytomegalovirus) are causative organisms.

Enteral Feeding The composition of tube feeding formula may cause diarrhea. High rate of feeding and high osmolality increase diarrhea incidence. Patients selected for tube feeding are often hypoalbuminemic and data suggest that hypoalbuminemia predisposes to diarrhea by decreasing osmotic pressure and causing edema in the intestinal mucosa. Whether fiber-containing formulas can control diarrhea due to tube feeding is unclear. Contamination of feeding and food poisoning is a common problem if hand washing hygiene, clean mixing of formula, refrigerating mixed formula, and proper handling of the feeding equipment are not observed.

Celiac Plexus Block Celiac plexus block is commonly associated with a self-limiting acute diarrhea; however, occasionally, it may be persistent [21]. This diarrhea may be amenable to treatment with atropine.

Stress and Anxiety The stress and anxiety associated with cancer and treatments can cause diarrhea [22].

MedicationsNotIntendedforCancerTreatment Excessive doses of laxatives or magnesium-containing antacids commonly result in diarrhea.

Symptoms

Increased frequency and volume of bowel movements per day, incontinence, increase in ostomy output volume compared with baseline, the character of the fecal material (loose,

Table 27.2 National Cancer Institute grading of diarrhea

very loose, watery), dizziness, abdominal pain, and fever are pieces of information needed for assessment of diarrhea severity [23]. These questions help classify diarrhea as complicated or uncomplicated and guide therapy [24]. Medication and dietary intake, as well as a history of recent travel, may provide additional etiologic clues. Weight loss and reduced urine output indicate the severity of diarrhea. Bloody stool, stool containing mucous, severe cramping, and abdominal pain are consistent with enterocolitis [25].

The National Cancer Institute Common Toxicity Criteria (version 4.03) is a frequently used standard tool for assessing diarrhea severity (Table 27.2) [26], but it does not include assessment of duration of diarrhea and stool volume. These severity parameters and other coexisting symptoms that are predictive of serious complications were addressed in the clinical practice guidelines [24, 27].

Grade 3 or 4 diarrhea is also classified as complicated. Diarrhea may be considered complicated and potentially serious if a cancer patient with Grade 1 or 2 diarrhea has the following: >6 loose bowel movements a day for >2 days, bloody stool or rectal bleeding, no urine output for >12 h, inability to drink for >1 day, weight loss due to diarrhea. diarrhea after several days of constipation, abdominal distension, or fever. In addition, moderate-to-severe cramping and nausea/vomiting, neutropenia, and the presence of systemic inflammatory response syndrome indicate potentially serious complications. Close monitoring and full investigation is warranted [24].

Diagnosis

Most chronic diarrheal paraneoplastic syndromes would have already been diagnosed in most cancer patients, and the diagnostic evaluation for the specific paraneoplastic syndromes is out of the scope of this chapter. For cancer patients presenting to EDs with diarrhea, the goal of rapid evaluation is to identify life-threatening conditions and complications requiring hospitalization. History should include medications (including antineoplastic agents), travel, and diet.

	Grade					
	0	1	2	3	4	5
Colostomy absent	None	Increase of <4 stools/day compared with baseline	Increase of 4–6 stools/day, or waking up to have bowel movements	Increase of \geq 7 stools/day or incontinence; or need for parenteral support for dehydration	Hemodynamic collapse or severe consequences requiring intensive care	Death
Colostomy present	None	Mild increase in loose watery output compared with baseline	Moderate increase in loose watery output compared with baseline, but not interfering with normal activity	Severe increase in loose watery output compared with baseline; or copious output that interferes with normal activity	Hemodynamic collapse or severe consequences requiring intensive care	Death

Adapted from National Institute of Health. National Cancer Institute. Common Toxicity Criteria (version 4.03) for Diarrhea https://nciterms.nci.nih.gov/ncitbrowser/ConceptReport.jsp?dictionary=CTCAE&ns=ctcae&code=E10575

Physical examination should assess intravascular volume status and ascertain signs of infection and abdominal tenderness.

Although a very crude estimate, information about the frequency of stools in the past 24 hours helps to assess severity of diarrhea. For immune-related diarrhea after immune checkpoint inhibitor therapy, Grade 1 (mild) diarrhea is up to four stools per day over baseline or mildly higher ostomy output from the baseline [13, 14, 16]; Grade 2 (moderate) diarrhea is four to six stools per day over baseline or moderate ostomy output; Grade 3–4 (severe or life-threatening) diarrhea is \geq seven stools per day over baseline, with fecal incontinence. The patient may present with symptoms such as abdominal pain, rectal bleeding, and mucus in stool [14], and bowel perforation can occur [16].

Diagnostic testing and imaging studies would follow, including routine complete blood count with differential and complete metabolic panel to assess leukocytosis or neutropenia, electrolyte imbalances, liver function, and renal function. Diarrhea causes dehydration, electrolyte abnormalities, and disturbed acid-base balance. Hypokalemia and nonanion gap acidosis are the main diagnostic features of severe diarrhea. Hypokalemia necessitates aggressive potassium replacement. Prerenal azotemia or renal failure may result from severe dehydration. Other electrolytes, including calcium and magnesium, should be checked and replaced. Hypokalemia, hypomagnesemia, and hypocalcemia can cause EKG changes and cardiac arrhythmias.

Stool samples should be analyzed by culture for the presence of pathogenic bacteria, such as Salmonella, Shigella, Campylobacter, Yersinia, C. difficile, and yeast; Clostridium difficile testing (enzyme immunoassay for C. difficile toxins A and B, C. difficile DNA PCR); and the presence of cryptosporidium by acid-fast stains and other parasites by standard laboratory methods. In the course of cancer treatment, many patients have infectious complications and exposure to antibiotics. Clostridium difficile-related diarrhea should always be excluded in cancer patients who have recently been treated with antibiotics. Viral pathogens can be tested by culture (enteroviruses and adenovirus), enzyme immunoassays (rotavirus and adenovirus), or multiplex PCR. Cytomegalovirus immediate-early antigen tests of blood can detect cytomegalovirus antigenemia.

Stool analysis can classify the diarrhea as watery, fatty, or inflammatory. Watery diarrhea suggests functional, secretory, or osmotic etiology. Functional disorders such as irritable bowel syndrome and functional diarrhea are common in chronic diarrhea. Secretory diarrhea can be caused by bile acid malabsorption, paraneoplastic syndromes, microscopic colitis, endocrine disorders, and some postsurgical changes. Osmotic diarrhea can be due to laxative abuse or malabsorption of nonlipid nutrients. Fatty diarrhea reflects malabsorption of fat, which can be caused by a wide variety of causes



Fig. 27.1 Pneumatosis intestinalis

such as digestive enzyme deficiency, disruption of biliary function, celiac disease, or ileitis. Inflammatory diarrhea warrants further evaluation and can be caused by inflammatory bowel diseases, immune colitis, and infectious enteritis [28]. Fecal lactoferrin and fecal calprotectin can help differentiate between an infectious or inflammatory etiology and can be used to monitor disease activity and treatment response [29].

An abdominal X-ray series is helpful to quickly exclude intraabdominal free air and pneumatosis intestinalis (Fig. 27.1). The suspicion for serious gastrointestinal complication (e.g., perforation and obstruction) should be high and the threshold for CT imaging of abdomen and pelvis should be low. A triad of neutropenia, abdominal tenderness, and diarrhea should raise suspicion for neutropenic enteritis [30]. Perhaps due to neutropenia, abdominal pain or tenderness may not be prominent despite the presence of significant infection. A CT scan of abdomen and pelvis with intravenous and oral contrasts can diagnose neutropenic ileitis, typhlitis, and colitis (Fig. 27.2). MRI may be used if CT is contraindicated (e.g., if the patient is allergic to iodine contrast dyes). Bedside ultrasonic examination of bowel wall thickness can provide a rapid diagnosis [31]. Thickening of bowel wall to >4 mm for >30 mm in length is suggestive of enterocolitis [27].

In cancer patients with significant diarrhea, colonoscopic examination and biopsy are often indicated. A history of allogeneic stem cell transplantation should put graft-versushost disease high in the differential diagnosis. Infectious colitis (e.g., cytomegalovirus colitis) is also in the differential diagnosis for immunocompromised patients. Immune colitis can also be caused by PI3K inhibitors. A history of ongoing treatment with immune checkpoint inhibition ther-



Fig. 27.2 CT scan imaging of a cancer patient with neutropenic ileitis. A neutropenic leukemic patient presents with profuse diarrhea and minimal abdominal pain but no fever. The CT scan demonstrated significant thickening of the wall of ileum (*arrows*) consistent with ileitis

apy should raise the suspicion for immune-mediated colitis, the confirmation of which would also require endoscopy, with or without biopsy. Viral colitis such as cytomegalovirus infection can also be confirmed by endoscopy with biopsies.

Management

Although many causes of diarrhea in cancer patients require specific therapies, therapies aimed at decreasing or replenishing fluid and electrolyte losses are required [32].

Diet

Regardless of the cause of diarrhea, diet modifications may decrease the symptom burden of diarrhea. Foods that may lessen the burden include the following:

- Six to eight small meals and snacks each day and increased room temperature clear liquids
- Low-fat high-potassium diet with foods containing soluble fiber
- Lactose-free
- BRAT (bananas, rice, applesauce, toast) diet

Foods that may worsen diarrhea should be avoided, such as:

- Fatty, greasy, or fried foods
- Foods high in insoluble fiber content
- · Gas-forming foods
- Foods with high sugar contents

- Hot liquids
- Dairy products or foods made with significant amount of dairy products
- Foods sweetened with sugar alcohols (e.g., sorbitol, xylitol, or mannitol)
- Foods that can irritate the digestive tract (e.g., caffeine such as coffee, strong tea, sodas, tomato juice, citrus juices, and alcohol)
- Tobacco

Probiotics before or during chemotherapy may prevent chemotherapy-induced diarrhea [33]. The use of probiotics appears helpful in improving tolerance of, and support for, treatment and radiation-related diarrhea. Sources of probiotics include foods such as yogurt, buttermilk, sauerkraut, and cottage cheese. Most clinical research involves *Lactobacillus* and *Bifidobacterium* to modify gut microflora [34]. However, probiotics are not recommended in immunocompromised neutropenic patients. Food is a potential cause of invasive infectious disease in immunocompromised patients, and *Lactobacillus acidophilus* bacteremia due to yogurt ingestion has been reported in a stem cell transplant patient with mucositis [35].

Medication Adjustment

Medications such as bulk laxatives, stool softener, and promotility agents (e.g., metoclopramide) should be discontinued. Oral magnesium supplements can cause diarrhea and if significant hypomagnesemia is present, parenteral magnesium replacement may be indicated.

Correction of Dehydration and Electrolyte Imbalances

Initial treatment for severe diarrhea is aimed at correcting any volume, electrolyte, and acid-base abnormalities with IV normal saline, potassium chloride, and if acidosis is severe, sodium bicarbonate. These abnormalities are frequently severe enough to necessitate hospital admission.

Pharmacologic Therapy

Treatment goals include slowing intestinal motility, decreasing intestinal secretions, and promoting intestinal absorption. Other pharmacologic therapies for the relief of diarrhea are specific to the underlying mechanism.

• Opioids bind to μ receptors in the gastrointestinal tract and decrease bowel motility to increase transit time:

- Loperamide; 4 mg followed by 2 mg after each unformed stool up to 12 mg/day [24, 27]
- Diphenoxylate
- Codeine
- Tincture of opium
- Anticholinergics:
 - Atropine
 - Belladonna
 - Scopolamine
- Adsorbents such as kaolin, clays, and activated charcoals have been commonly used and generally considered safe:
 - Kaolin
 - Pectin
- Absorbents give bulk to the fecal material, but one potential drug interaction is that they may bind and inhibit absorption of other oral antidiarrheal medications:
 - Wheat dextrin
 - Psyllium fiber
- Somatostatin analogues:
 - Octreotide; treatment usually start with 100–150 μg every 8 hours
 - Lanreotide
 - Pasireotide
- Mucosal prostaglandin inhibitors has antisecretory effects:
 - Aspirin (may be useful for radiation-induced diarrhea)
 - Bismuth subsalicylate
- Corticosteroids reduce edema associated with obstruction and radiation colitis, reduce hormonal influences of some endocrine tumors (e.g., VIPoma), and treat immunemediated colitis:
 - Budesonide is an oral steroid medication that is topically active in the gastrointestinal tract. It has a 90% first pass effect; therefore, after liver metabolism, the systemic availability is low
 - Dexamethasone
 - Methylprednisolone, prednisolone, or prednisone
- Antimicrobials: Quinolone antibiotics are effective for salmonellosis. Depending on the degree of immunocompromise, antibiotic treatment may need to continue for several months. Some beta-lactam antibiotics (e.g., cefotaxime, ceftriaxone) and sulfamethoxazole-trimethoprim are alternatives. Campylobacteriosis is treated with azithromycin or quinolone antibiotics, with addition of vancomycin for severe cases. Shigellosis is treated with quinolone antibiotics. Alternatives include sulfamethoxazole-trimethoprim and azithromycin. Clostridium difficile may be treated with metronidazole or oral vancomycin. Enterotoxigenic E. coli are frequently resistant to ampicillin and sulfamethoxazole-trimethoprim. Quinolone antibiotics are generally effective. Enterotoxigenic Bacteroides fragilis is an emerging

pathogen causing diarrhea, and metronidazole has excellent activity against this pathogen

- Bismuth subsalicylate has direct antimicrobial effects on *Escherichia coli*
- Sulfamethoxazole and trimethoprim
- Beta-lactam antibiotics, e.g., cefotaxime, ceftriaxone
- Quinolone antibiotics, e.g., levofloxacin, moxifloxacin, ciprofloxacin
- Metronidazole
- Oral vancomycin

Management of Specific Clinical Scenarios

Treatment-Induced Diarrhea

Based on controlled clinical trials and clinical practice guidelines [24, 27], loperamide (4 mg initial dose followed by 2 mg every 4 hours) is the standard first-line therapy for chemotherapy-induced diarrhea. After loperamide for the first day of chemotherapy-induced mild diarrhea, treatment may be escalated by adding octreotide, 100–150 µg every 8 hours [23]. Severe treatment-induced diarrhea with complicated symptoms should be managed with IV fluids, octreotide acetate 100–150 µg SC three times daily or 25–50 µg/hour IV with up to a five-fold escalation as needed, and administration of antibiotics until diarrhea has stopped for >24 hours [24]. Updated guidelines stress the importance of recognizing early warning signs of complicated diarrhea and early intervention such as initiating antibiotic therapy [24].

"Complicated" Chemotherapy-Induced Diarrhea

Patients with mild-to-moderate diarrhea complicated by moderate-to-severe cramping, nausea and vomiting, diminished performance status, fever, sepsis, neutropenia, bleeding, or dehydration, as well as patients with severe diarrhea, are classified as "complicated." These patients should be evaluated further, monitored closely, and treated aggressively. Aggressive management of complicated cases usually necessitates hospital admission and involves IV fluids; octreotide at a starting dose of 100–150 µg three times daily $(25-50 \mu g/h)$ if the patient is severely dehydrated, with dose escalation up to 500 µg three times daily until diarrhea is controlled, and administration of antibiotics (e.g., fluoroquinolone). These patients should be evaluated with complete blood count, electrolyte profile, and a stool work-up evaluating for blood, fecal leukocytes, C. difficile, Salmonella, E. coli, Campylobacter, and infectious colitis.

Targeted Therapy

There are no specific guidelines for the management of targeted therapy-induced diarrhea. As a secretive mechanism is the most frequent cause, the first-line treatment is loperamide and opiates, followed by octreotide.

Neutropenic Enterocolitis

The risk of mortality from neutropenic enterocolitis is high [30, 36]. In the absence of acute complications that require emergency surgery, the initial management is to use broad-spectrum antibiotics. The antibiotics chosen should cover enteric gram-negative organisms, gram-positive organisms, and anaerobes. The choice of antibiotics is basically the same as for neutropenic fever. In cases that do not respond to antibiotics, antifungal agents should be added, just as in the management of neutropenic fever. Depending on the underlying malignancy, neutropenia may not resolve quickly, even when colony-stimulating factors (filgrastim) are used.

Immune-Mediated Colitis

There are quite a few guidelines available for the management of immune-related adverse effects of immune checkpoint inhibitors, including those from the National Comprehensive Cancer Network (NCCN) [37], American Society of Clinical Oncologic (ASCO) [38], Society of Immunotherapy for Cancer (SITC) [39], and European Society of Medical Oncologic (ESMO) [40]. Management is based on grading. Grade 1: Symptomatic treatment with loperamide, oral hydration, the American Dietary Association ulcerative colitis diet, and electrolyte replacement as needed. Grade 2: Diphenoxylate/atropine may replace loperamide and budesonide (or systemically active prednisone/methylprednisolone) may be started. In grade 1 or 2 diarrhea with bleeding or persistent grade 2 diarrhea, endoscopic examination and biopsy are indicated. Grade 3 or 4: Treatment with IV corticosteroid (methylprednisolone 125 mg) and IV replacement of fluid and electrolytes should be started. Infliximab is second-line treatment and is usually started for the lack of improvement in diarrhea after treating with corticosteroids for 72 hours. Corticosteroids are typically the first-line treatment for irAE grade 2 or higher, and the primary therapy that emergency physicians will institute.

GVHD

In addition to antidiarrheal agents, glucocorticoids (e.g., budesonide), and immunosuppressive medications, the diarrhea associated with GVHD may be managed with a specialized five-phase dietary regimen [41]. Octreotide is also effective in diarrhea associated with GVHD [42, 43].

Paraneoplastic Diarrhea

Initial treatment is directed toward correcting volume and electrolyte abnormalities. Somatostatin analogues control diarrhea in up to 90% of patients. Glucocorticoids reduce symptoms in 50% of cases. Tumor resection is the treatment of choice for long-term control of symptoms. In advanced disease, tumor debulking may relieve symptoms, but it is not effective in all cases. Hepatic artery radioembolization or transcatheter chemoembolization with doxorubicin or cisplatin [44], XRT, and percutaneous or intraoperative radiofrequency tumor ablation may be attempted to reduce tumor burden.

References

- 1. Arbuckle RB, Huber SL, Zacker C. The consequences of diarrhea occurring during chemotherapy for colorectal cancer: a retrospective study. Oncologist. 2000;5(3):250–9.
- Abu-Sbeih H, Mallepally N, Goldstein R, Chen E, Tang T, Dike UK, et al. Gastrointestinal toxic effects in patients with cancer receiving platinum-based therapy. J Cancer. 2020;11(11):3144–50.
- Chen E, Abu-Sbeih H, Thirumurthi S, Mallepally N, Khurana S, Wei D, et al. Clinical characteristics of colitis induced by taxanebased chemotherapy. Ann Gastroenterol. 2020;33(1):59–67.
- Pessi MA, Zilembo N, Haspinger ER, Molino L, Di Cosimo S, Garassino M, et al. Targeted therapy-induced diarrhea: A review of the literature. Crit Rev Oncol Hematol. 2014;90(2):165–79.
- Chen Z, Wei J, Ma X, Yu J. Efficacy of EGFR-TKIs with or without angiogenesis inhibitors in advanced non-small-cell lung cancer: A systematic review and meta-analysis. J Cancer. 2020;11(3):686–95.
- Shohdy KS, Lasheen S, Kassem L, Abdel-Rahman O. Gastrointestinal adverse effects of cyclin-dependent kinase 4 and 6 inhibitors in breast cancer patients: a systematic review and meta-analysis. Ther Adv Drug Saf. 2017;8(11):337–47.
- 7. Nunnery SE, Mayer IA. Management of toxicity to isoform α-specific PI3K inhibitors. Ann Oncol. 2019;30(Suppl_10):x21–6.
- Greenwell IB, Ip A, Cohen JB. PI3K Inhibitors: Understanding Toxicity Mechanisms and Management. Oncology (Williston Park). 2017;31(11):821–8.
- Sipaviciute A, Sileika E, Burneckis A, Dulskas A. Late gastrointestinal toxicity after radiotherapy for rectal cancer: a systematic review. Int J Color Dis. 2020;35(6):977–83.
- Wang L, Zhou Y, Wang X, Zhang G, Guo B, Hou X, et al. Mechanism of Asbt (Slc10a2)-related bile acid malabsorption in diarrhea after pelvic radiation. Int J Radiat Biol. 2020;96(4):510–9.
- 11. Arango JI, Restrepo A, Schneider DL, Callander NS, Ochoa-Bayona JL, Restrepo MI, et al. Incidence of *Clostridium difficile*associated diarrhea before and after autologous peripheral blood stem cell transplantation for lymphoma and multiple myeloma. Bone Marrow Transplant. 2006;37(5):517–21.
- Robak K, Zambonelli J, Bilinski J, Basak GW. Diarrhea after allogeneic stem cell transplantation: beyond graft-versus-host disease. Eur J Gastroenterol Hepatol. 2017;29(5):495–502.
- Lomax AJ, McNeil C. Acute management of autoimmune toxicity in cancer patients on immunotherapy: Common toxicities and

the approach for the emergency physician. Emerg Med Australas. 2017;29(2):245–51.

- Rajha E, Chaftari P, Kamal M, Maamari J, Chaftari C, Yeung SJ. Gastrointestinal adverse events associated with immune checkpoint inhibitor therapy. Gastroenterol Rep (Oxf). 2020;8(1):25–30.
- Bellaguarda E, Hanauer S. Checkpoint inhibitor-induced colitis. Am J Gastroenterol. 2020;115(2):202–10.
- Kumar V, Chaudhary N, Garg M, Floudas CS, Soni P, Chandra AB. Current diagnosis and management of immune related adverse events (IRAES) induced by immune checkpoint inhibitor therapy. Front Pharmacol. 2017;8:49.
- Abu-Sbeih H, Tang T, Lu Y, Thirumurthi S, Altan M, Jazaeri AA, et al. Clinical characteristics and outcomes of immune checkpoint inhibitor-induced pancreatic injury. J Immunother Cancer. 2019;7(1):31.
- Brahmer JR, Lacchetti C, Schneider BJ, Atkins MB, Brassil KJ, Caterino JM, et al. Management of Immune-Related Adverse Events in Patients Treated With Immune Checkpoint Inhibitor Therapy: American Society of Clinical Oncology Clinical Practice Guideline. J Clin Oncol. 2018;36(17):1714–68.
- Cramer P, Bresalier RS. Gastrointestinal and hepatic complications of immune checkpoint inhibitors. Curr Gastroenterol Rep. 2017;19(1):3.
- Ikeuchi K, Okuma Y, Tabata T. Immune-related pancreatitis secondary to nivolumab in a patient with recurrent lung adenocarcinoma: A case report. Lung Cancer. 2016;99:148–50.
- Yang A, Brown J, Mak E. Persistent diarrhea after celiac plexus block in a pancreatic cancer patient: case report and literature review. J Palliat Med. 2016;19(1):83–6.
- Mercadante S. Diarrhea in terminally ill patients: pathophysiology and treatment. J Pain Symptom Manag. 1995;10(4):298–309.
- Kornblau S, Benson AB, Catalano R, Champlin RE, Engelking C, Field M, et al. Management of cancer treatment-related diarrhea. Issues and therapeutic strategies. J Pain Symtpom Manage. 2000;19(2):118–29.
- Benson AB 3rd, Ajani JA, Catalano RB, Engelking C, Kornblau SM, Martenson JA Jr, et al. Recommended guidelines for the treatment of cancer treatment-induced diarrhea. J Clin Oncol. 2004;22(14):2918–26.
- Thompson JA, Schneider BJ, Brahmer J, Andrews S, Armand P, Bhatia S, et al. NCCN Guidelines insights: Management of immunotherapy-related toxicities, Version 1.2020. J Natl Compr Cancer Netw. 2020;18(3):230–41.
- National Cancer Institute. Cancer Therapy Evaluation Program (CTEP). Common toxicity criteria (CTC). Version 2.0. Revised 23 Mar 2998; Published 30 Apr 1999. http://ctep.cancer.gov/Forms/ CTCv20_4-30-992.pdf. Accessed 10 Nov 2020.
- 27. Wadler S, Benson AB 3rd, Engelking C, Catalano R, Field M, Kornblau SM, et al. Recommended guidelines for the treatment of chemotherapy-induced diarrhea. J Clin Oncol. 1998;16(9):3169–78.
- Burgers K, Lindberg B, Bevis ZJ. Chronic diarrhea in adults: evaluation and differential diagnosis. Am Fam Physician. 2020;101(8):472–80.

- Reddy HG, Schneider BJ, Tai AW. Immune checkpoint inhibitorassociated colitis and hepatitis. Clin Transl Gastroenterol. 2018;9(9):180.
- Gorschluter M, Mey U, Strehl J, Ziske C, Schepke M, Schmidt-Wolf IG, et al. Neutropenic enterocolitis in adults: systematic analysis of evidence quality. Eur J Hematol. 2005;75(1):1–13.
- Tamburrini S, Setola FR, Belfiore MP, Saturnino PP, Della Casa MG, Sarti G, et al. Ultrasound diagnosis of typhlitis. J Ultrasound. 2019;22(1):103–6.
- 32. Cherny NI. Evaluation and management of treatment-related diarrhea in patients with advanced cancer: a review. J Pain Symptom Manag. 2008;36(4):413–23.
- 33. Lu D, Yan J, Liu F, Ding P, Chen B, Lu Y, et al. Probiotics in preventing and treating chemotherapy-induced diarrhea: a metaanalysis. Asia Pac J Clin Nutr. 2019;28(4):701–10.
- 34. Isolauri E. Probiotics in human disease. Am J Clin Nutr. 2001;73(6):1142S–6S.
- 35. Mehta A, Rangarajan S, Borate U. A cautionary tale for probiotic use in hematopoietic SCT patients-*Lactobacillus acidophilus* sepsis in a patient with mantle cell lymphoma undergoing hematopoietic SCT. Bone Marrow Transplant. 2013;48(3):461–2.
- Rodrigues FG, Dasilva G, Wexner SD. Neutropenic enterocolitis. World J Gastroenterol. 2017;23(1):42–7.
- Thompson JA. New NCCN guidelines: recognition and management of immunotherapy-related toxicity. J Natl Compr Cancer Netw. 2018;16(5S):594–6.
- Brahmer JR, Lacchetti C, Schneider BJ, Atkins MB, Brassil KJ, Caterino JM, et al. Management of Immune-Related Adverse Events in Patients Treated With Immune Checkpoint Inhibitor Therapy: American Society of Clinical Oncology Clinical Practice Guideline. J Clin Oncol. 2018;36(17):1714–68.
- 39. Puzanov I, Diab A, Abdallah K, Bingham CO, Brogdon C, Dadu R, et al. Managing Toxicities Associated With Immune Checkpoint Inhibitors: consensus recommendations from the Society for Immunotherapy of Cancer (SITC) Toxicity Management Working Group. J Immunother Cancer. 2017;5(1):95.
- 40. Haanen JBAG, Carbonnel F, Robert C, Kerr KM, Peters S, Larkin J, et al. Management of Toxicities From Immunotherapy: ESMO Clinical Practice Guidelines for Diagnosis, Treatment and Follow-Up. Ann Oncol. 2017;28(Suppl 4):iv119–iv42.
- Charuhas PM. Medical nutrition therapy in bone marrow transplantation. In: McCallum PD, Polisena CG, editors. The clinical guide to oncology nutrition. Chicago: The American Dietetic Association; 2000. p. 90–8.
- 42. Ippoliti C, Champlin R, Bugazia N, Przepiorka D, Neumann J, Giralt S, et al. Use of octreotide in the symptomatic management of diarrhea induced by graft-versus-host disease in patients with hematologic malignancies. J Clin Oncol. 1997;15(11):3350–4.
- Morton AJ, Durrant ST. Efficacy of octreotide in controlling refractory diarrhea following bone marrow transplantation. Clin Transpl. 1995;9(3 Pt 1):205–8.
- 44. Khasraw M, Gill A, Harrington T, Pavlakis N, Modlin I. Management of advanced neuroendocrine tumors with hepatic metastasis. J Clin Gastroenterol. 2009;43(9):838–47.



Constipation

Demis N. Lipe

28

Case Study

An 82-year-old male with pancreatic cancer presents to the emergency department with complaints of constipation for 5 days. He has tried over-the-counter laxatives without relief. He is nauseated, but not vomiting, and describes a sensation of pressure in his suprapubic region. He can pass gas but is having difficulty urinating. He has not had any recent medication changes. His cancer-associated pain is well controlled by his daily opioid regimen. He has a past history of constipation. On presentation, he is somewhat cachectic and distressed. His vitals are normal, and physical examination reveals a mildly tender lower abdomen, without signs of peritonitis. His complete blood count, chemistries, and urinalysis reveal his baseline anemia of chronic disease, but otherwise, his laboratory findings are normal. A plain radiograph (Fig. 28.1) shows a large stool burden, but no evidence of free air or dilated loops of bowel. Rectal exam reveals hard stool in the vault, but no evidence of stool impaction. The patient is gently hydrated and treated with a soapsuds enema, which successfully leads to a bowel movement in the emergency department and relief of symptoms. He is discharged home uneventfully.

Introduction

Constipation is a common gastrointestinal symptom that varies widely in severity. Although definitions vary, constipation is generally characterized by the slow movement of feces through the intestines with fewer bowel movements than normal [1, 2]. Constipation may be transient and easily treated or cause major impairment and be resistant to multiple interventions, among both the general population and

Department of Emergency Medicine, Division of Internal Medicine, The University of Texas MD Anderson Cancer Center, Houston, TX, USA e-mail: dnlipe@mdanderson.org



Fig. 28.1 Large amount of stool in upright abdominal radiograph

those with cancer. It is often underappreciated and undertreated in patients with cancer, and it has been reported to affect up to 50% of patients with advanced cancer [3].

Its reported prevalence varies from 1% to 81%. Such wide variation results from nonspecific definitions and differing base populations [2, 4–8]. More specifically, the prevalence of constipation among cancer patients ranges from 30-90%, and increases with age, opioid therapy, hospitalization, and need for palliative care [1, 9]. Age-related increases in constipation occur in the setting of immobility, dehydration, polypharmacy, and with advanced age, a reduced urge to defecate [1]. Among cancer patients receiving palliative care, those under 60 years of age reported a 33% prevalence of constipation, rising to 55% among those 70 years or older [10]. In one cross-sectional study of 520 cancer patients receiving strong opioids, 321 (61.7%) patients were assessed as constipated using a validated constipation questionnaire, while by physicians' subjectively judgment, 438 (85.7%) were constipated [11].

D. N. Lipe (🖂)

Emergency department (ED) visits related to constipation are common both in the general and cancer population. Highlighting the healthcare burden of constipation, Sommers et al. found that of 131,048,605 total US ED visits in 2011, over 700,000 were related to constipation [12]. In 2019, of 28,424 total patient visits to the MD Anderson Cancer Center ED in Houston, Texas, 15,289 (54%) visits included a diagnosis related to constipation (unpublished data).

ED visits for constipation entail large and increasing healthcare expenditures. US ED visits for constipation increased by 41.5% between 2006 and 2011, from 497,034 to 703,391 visits, while the national cost of constipation visits increased from \$733 million to \$1.6 billion (in 2014 dollars) [12]. In the United Kingdom, cost to the National Health Service for treatment of constipation in the year 2017–2018 was estimated to be 162 million pounds and on average, 196 people were admitted to the hospital daily during the same time period. In addition, millions of dollars are spent annually on laxatives and primary care visits [8, 13–15].

In addition to healthcare costs, constipation-related symptoms are distressing and debilitating for those with cancer. Stigma and shame may prevent the patient from disclosing constipation symptoms to loved ones or healthcare providers [8, 13].

The Rome Diagnostic Criteria

Due to lack of overarching operational definitions for functional bowel disorders, an international group of gastroenterologists convened in 1988 in Rome, Italy to develop consensus criteria for the diagnosis of irritable bowel syndrome. This led to further attempts to develop systematic nomenclatures for a number of symptom-based functional gastrointestinal (GI) disorders. With an expanding research base, these taxonomical efforts were increasingly supported by physiologic data [16]. To date, Rome diagnostic criteria have become the gold standard for defining a number of functional GI disorders, including constipation. Despite accepted Rome criteria, there is limited agreement between generalist clinicians, expert specialists, and the public in recognizing and differentiating constipation from other functional GI disorders and nonpathologic symptoms. Experts recognize the need to align Rome diagnostic criteria with those symptoms that patients perceive to be important and functionally disabling [17].

The Rome diagnostic criteria categorize functional bowel disorders into five syndromes: irritable bowel syndrome, functional constipation, functional diarrhea, functional abdominal bloating/distention, and unspecified functional bowel disorder. Functional constipation is the most common form of constipation. A diagnosis of functional constipation per the Rome IV criteria requires that

 Table 28.1
 Rome IV criteria for functional constipation [18]

Duration of 3 months and symptom onset at least 6 months prior to							
d	diagnosis						
1		Presence of ≥ 2 of the following:					
		Lumpy or hard stools in >25% of defecations					
		Straining during >25% of defecations					
		Sensation of anorectal obstruction/blockage for >25% of					
		defecations					
		Sensation of incomplete evacuation for >25% of defecations					
		Using manual maneuvers such as digital manipulation or					
		pelvic floor support to facilitate >25% of defecations					
		<3 spontaneous bowel movements per week					
		and					
2		Loose stools rarely present without the use of laxatives					
		and					
3		Insufficient criteria for irritable bowel syndrome					

symptoms must be present for the last 3 months, with an overall onset of symptoms 6 months prior to diagnosis (Table 28.1). The Rome IV criteria include a new sixth category, opioid-induced constipation (OIC), particularly relevant to cancer patients [18]. OIC is defined as new or worsening constipation when initiating, changing, or increasing opioid therapy. While OIC is considered an adverse effect of opioids rather than a functional bowel disorder, the two often coexist. To meet OIC criteria, patients must experience two or more of the symptoms that define functional constipation with the same frequency cut-off of 25%. The Rome IV criteria also address narcotic bowel syndrome (NBS). NBS was first described over 25 years ago as a chronic or frequently occurring paradoxical development of, or increase in, abdominal pain associated with increasing doses of opioids [18–21].

Pathophysiology

Constipation can be attributed to primary causes such as colonic or anorectal dysfunction or secondary causes such as organic disease or medication use [20]. A number of factors are associated with constipation, including, but not limited to, dietary habits, genetic predisposition, absorption ability, colonic motility, daily behaviors, medication use, spinal cord compression or injury, musculoskeletal disorders such as muscular dystrophy, and external colonic compression (e.g., peritoneal tumors) [7].

Secondary constipation is often adverse effect of chemotherapeutic agents and medications used to treat concomitant diseases and symptoms. Constipation has been cited as the third most common symptom associated with chemotherapy administration [22]. A number of other drug classes cause constipation by reducing smooth muscle contractility (e.g., calcium channel antagonists, antihistamines, antidepressants) [7, 20]. Refer to Table 28.2 for a more comprehensive list of medications causing constipation.

Drug class	Example of drug
Alpha-2 agonists	Clonidine, methyldopa
Calcium channel blockers	Verapamil, nifedipine
Tricyclic antidepressants	Doxepin, amitriptyline
Metal ion-containing agents	Iron, aluminum, bismuth, calcium
Opioids	Morphine, fentanyl,
	hydromorphone
Cannabinoids	Tetrahydrocannabinol (THC)
Antiulcer agents	Sucralfate, antacids
Anti-Parkinson drugs	Methyldopa
Antiepileptics	Phenytoin, carbamazepine
Antipsychotics	Clozapine, olanzapine, quetiapine
Antihistamines	Diphenhydramine, cetirizine
Antispasmodics	Dicyclomine and hyoscyamine
Anticholinergics	Belladonna, rasagiline
5-HT3 receptor antagonist	Ondansetron
Chemotherapeutic agents	Cyclophosphamide, vincristine
Nonsteroidal	Toradol, ibuprofen
anti-inflammatories	
Bile acid sequestrants	Cholestyramine

	- 20 1	1	1	• .•	•		
1201		 ommon	mod	109f10nc	concing	concti	nation
IGNI	C 20.	ommon	meu	ications.	Causing	COnsu	Dauon

Chemotherapy-induced constipation is not well understood, but it is believed that it stems from the effects of the chemotherapeutic agent on nerve endings of the GI enteric nervous system. It is suggested that subtle changes in the enteric nervous system lead to abnormal colonic motor function, prolonged colonic transit time, infrequent bowel movements, and thus, constipation [22].

Opioids reduce bowel motility by binding to specific μ , δ , and κ receptors in the GI tract. Under normal circumstances, these receptors influence fluid and electrolyte transport in the stomach, ileum, and proximal colon, as well as promoting motility. Opioids bind to the μ receptor and lead to a decrease in peristaltic activity as well as reduction in mucosal secretions, ultimately causing decreased intestinal fluid absorption, slowed gastric emptying, and delayed intestinal traffic. Opioids also can weaken the susceptibility of rectal dilation to stimulation, reducing contractility as well as increasing sphincter tone [7, 22–25].

Narcotic bowel syndrome (NBS) is an entity deserving special mention as emergency physicians treating cancer patients may frequently encounter this condition. Pain is one of the most common complaints in the cancer population presenting to the ED and up to 60% of those patients are treated with opioids [26]. With continued opioid use, a paradoxical worsening of abdominal pain can occur despite increased opioid dose. Although the pathophysiology is not fully understood, NBS is thought to be related to heightened pain sensitivity localized to the GI tract. Chronic opioid use can cause neuronal changes that may lead to hyperalgesia. Other mechanisms include interactions with *N*-methyl D-aspartate receptor at the level of the spinal cord, increase in substance P synthesis, interaction with transmembrane G proteins, which have both inhibitory and excitatory effects

on sensory neurons, increase in the endogenous opioid peptide dynorphin, which in turn leads to increased pain signaling at the level of the spinal cord, and neuroinflammation resulting in hyperalgesia [21, 27].

Finally, spinal cord compression or injury may cause spasticity in the muscles of the GI tract. Neurogenic bowel dysfunction can be characterized by increased colonic and anal sphincter tone if the compression or injury is above the conus medullaris [28].

Clinical Presentation

Cancer patients with constipation present with an array of symptoms. Given the range of pathophysiology associated with cancer and its treatment, it is imperative that the emergency physician not overlook a more urgent condition that may mimic constipation. The history of present illness should include duration of symptoms, history of similar symptoms, consistency and frequency of stools, sensation of blockage or a feeling of incomplete evacuation, straining, recent dietary changes and recent medication changes, or increase in opioids. A thorough past medical history, to include comorbidities such as diabetes, diverticulitis, past surgeries, and current medication usage, should be elicited [7]. A recent increase in cancer pain, requiring increase in opioids is often an important aspect of the history, thus the pain level should also be recorded.

Patients with NBS may describe a burning or colicky pain that is exacerbated by eating, progressive, and associated with nausea, vomiting, abdominal distention, and bloating [21, 27]. Cancer patients presenting to the ED with NBS are often difficult to diagnose and consultation with their oncologist or continuity physician may be helpful. The diagnostic criteria for NBS are shown in Table 28.3.

Symptoms of abdominal pain, bloating, nausea, inability to pass stool, small stools, straining, rectal pain, urinary retention, and overflow diarrhea are commonly present. More alarming findings such as change in bowel habits, weight loss, anemia, not passing flatus, vomiting, bloody

 Table 28.3
 Diagnostic criteria for narcotic bowel syndrome [21]

 Chronic or frequently recurring abdominal pain that is treated with high dose opioids (acutely or chronically)
 and
 Presence of at least three of the following: The pain worsens or incompletely resolves with continued or increasing doses of opioids
 There is worsening of pain when the opioid dose wanes and

improvement when the opioids are reinstituted

There is progression of the frequency, duration, and intensity of pain episodes

The nature and intensity of the pain is not explained by a current or previous GI diagnosis

stools, history of inflammatory bowel disease, or history of colon cancer should prompt an expedited workup. Lower extremity weakness or numbness with urinary retention or in the setting of constipation should be further investigated to ensure there is no cord compression.

The Bowel Function Index three-item tool can be used for screening for opioid-induced constipation. This tool can be administered by the physician at the bedside and it consists of the patient's assessment of easiness of defecation in the last 7 days, the feeling of incomplete bowel evacuation during the last 7 days, and the patient's subjective judgment about the presence of constipation during the last 7 days [29].

Examination

Examination should begin with an evaluation for systemic toxicity. Does the patient appear acutely ill? Does the patient need acute resuscitation? It should then be followed by careful inspection of the abdomen, taking notice of the contour and movements of the abdomen as well as observing for symmetry. Inspection should be followed by auscultation, which can provide information about bowel motility. It is important to listen before palpating the abdomen as palpation can alter the frequency of bowel sounds. Next, palpating the abdomen will help with distinguishing masses, stool, a distended bladder, or signs of peritoneal inflammation. A digital rectal examination may be performed to check for stool impaction and sphincter tone. A history of bloody stools should prompt an examination for hemorrhoids or anal fissures. However, rectal exam in the oncologic population should be postponed until there is evidence that the patient is not neutropenic.

After a focused examination of the abdomen, a more thorough examination may be done in search of other systemic causes of constipation, such a hypothyroidism, hypercalcemia, spinal cord compression, and dehydration.

Differential Diagnosis

The first step in the diagnostic approach is to form a broad differential to ensure more serious causes are not missed. Differential diagnosis possibilities should include the following:

1. *Bowel Obstruction:* Bowel obstruction (small or large) can be caused by scar tissue, hernias, and malignancy. Patients presenting with bowel obstruction can present with severe pain, fever, bilious or nonbilious vomiting, bloating, and obstipation. However, patients can also present with variable, nonclassical symptoms. Having a history of mass in the GI tract, prior surgery, or abdominal hernias may further point to a bowel obstruction. Clinically, patients may appear ill, have a distended abdomen that is often tympanic with decreased or increased bowel sounds, and may be diffusely tender with or without signs of peritoneal irritability. In this case, a supine and upright radiograph may demonstrate distended loops of bowel or air-fluid levels indicating possible obstruction. A CT of the abdomen should be obtained to better evaluate and delineate the cause of obstruction [30]. If the diagnosis is missed, bowel obstructions can progress and cause bowel ischemia.

2. Bowel Perforation: Bowel perforation is a surgical emergency. There are multiple etiologies, but commonly neoplasm, diverticulitis, abscesses, inflammatory and infectious colitis, as well as bowel ischemia should be considered as causes of perforation. Stercoral colitis (Figs. 28.2 and 28.3) is a rare complication due to fecal impaction leading to increased intraluminal wall pressure, inflammation, and distention of the colon, which in turn leads to ischemic pressure and necrosis with ulceration and perforation. It can result in peritonitis and rapidly progress to septic shock. Neutropenic colitis may also cause transmural necrosis and perforation if severe. These patients will often present with severe pain, abdominal distention, nausea, and/or vomiting and possibly obstipation or diarrhea, depending on the etiology. On physical examination, the patient appears ill and may exhibit findings consistent with peritonitis. Workup



Fig. 28.2 Plain radiograph demonstrating air-fluid levels and a distended bowel in a patient with impacted fecal material



Fig. 28.3 Stercoral colitis in a patient with impacted fecal material

should include an emergent upright chest radiograph to evaluate for free air under the diaphragm. However, if plain radiography is negative, a CT of the abdomen should be immediately obtained, as plain radiographs have only a sensitivity of 50–70% in detecting extraluminal air [31].

3. Malignant Spinal Cord Compression: Malignant cord compression is an oncologic emergency and a potentially devastating condition, that if missed can lead to catastrophic neurological deficits. The condition has been reported to occur in approximately 5% of all cancer patients and up to 20% of the time, it is the initial manifestation of cancer [32]. Its incidence in advanced cancer has been reported to be approximately 15% [33]. Constipation may be the initial presenting symptom. A thorough history of present illness and review of systems may allow an early diagnosis. Neurological symptoms may include extremity weakness, numbness, and tingling. Symptoms such as urinary retention or overflow inconti-

nence are also important red flags. Known bony metastatic disease increase the likelihood of cord compression and thorough questioning about back pain and changes in pain intensity may be helpful. The neurological exam may be normal in the early stages of cord compression. Rectal tone should be checked, if the patient is not neutropenic. If malignant spinal cord compression is suspected, an emergent magnetic resonance imaging study of the spine is indicated with concomitant initiation of systemic steroids.

- 4. Diverticulitis: This condition can often present subtly, but if missed can cause complications such as phlegmon formation and perforation. The patients may present with fever and lower abdominal pain. Nausea and/or vomiting may be present or absent and the patient may give a history of diarrhea or constipation. Exam will often show tenderness of the left lower quadrant and the diagnosis can be confirmed with CT scan of the abdomen. The patient may or may not have an associated leukocytosis.
- 5. Hypothyroidism: While most cases of hypothyroidism are due to a primary process in the thyroid, a small percentage are due to other causes, such as neoplasm and medications (e.g., chemotherapeutic agents and immunotherapy). Patients may present with subtle symptoms including constipation, fatigue, muscle cramps, weight gain, voice changes, dry skin, and cold sensitivity. In these patients, again a thorough history is important as is a detailed review of systems. Exam may reveal a goiter, dry skin, pretibial edema, and delayed relaxation of deep tendon reflexes. Checking a serum thyrotropin level (TSH) is the single best screening test for primary thyroid dysfunction, but not adequate for hospitalized or acute ill patients [34]. Additional testing of for free thyroxine levels (T4) as well as triiodothyronine (T3) may assist the diagnosis.
- 6. Hypercalcemia of Malignancy: One of the most common causes of hypercalcemia is malignancy, affecting approximately 30% of patients with cancer at some point. Hypercalcemia may be due to bony metastasis or secretion of parathyroid hormone-related protein. Cancers with higher risk of hypercalcemia are multiple myeloma, renal cell carcinoma, lung, breast, and colorectal cancers. Drugs such as tamoxifen can also cause hypercalcemia. The patient may present with nonspecific complaints such as fatigue, anorexia, constipation, nausea, and abdominal pain. They may also present with pancreatitis and exhibit an elevated lipase, or nephrolithiasis with renal insufficiency. Clinically, the patient may appear dehydrated, lethargic, or even confused, but more often the physical exam is nonspecific. High blood pressure, bony tenderness, lower extremity muscle weakness, and tongue fasciculations may be seen. Workup should include chemistries with ionized calcium as well as liver

enzymes to include albumin, as up to 40% of calcium is protein bound. In this case, the constipation is secondary to the primary disease. An EKG should be obtained to check for QT interval shortening and Osborn or J waves and intravenous fluids initiated in the absence of contraindications [35].

Diagnosis

Assuming an adequate differential has been considered, constipation can be diagnosed simply by history and exam. In fact, the American College of Gastroenterology Chronic Constipation Task Force recommends against routine diagnostic testing in chronic constipation if there are no red flags [36]. However, the presence of cancer will often prompt further diagnostic testing.

In the ED, a complete blood count should be ordered to evaluate for neutropenia prior to performing a rectal exam. Chemistries to include electrolytes should be done to exclude metabolic causes of constipation. Thyroid function tests may be helpful if hypothyroidism is suspected. While imaging is not routinely required in the ED when uncomplicated constipation is suspected, an acute abdominal series radiograph with supine and upright views may reveal free air and airfluid levels. If there are any red flags in the history suggesting a more serious process, computed tomography of the abdomen should be performed.

Treatment

Once it is determined that the patient has uncomplicated constipation and other emergent causes (e.g., bowel obstruction, spinal cord compression) ruled out, the next step prior to treatment is to determine whether constipation is primary or secondary. Primary constipation is due to intrinsic defects in bowel function and usually is considered only after secondary causes of constipation have been ruled out [20]. The cancer patient is likely to have constipation due to secondary causes, such as medication use, or the malignancy itself.

As constipation is often multifactorial, the treatments will vary. In general, multiple approaches should be employed. These begin with hydration of the patient if the clinical picture warrants it, as well as correction of abnormal electrolytes. Most cancer patients presenting to the ED have already attempted oral medications such as osmotic or stimulant laxatives that have failed to relieve the symptoms. Assuming home treatment has failed, the most common approach in the ED is the administration of enemas and suppositories, followed by oral laxatives. Another less appealing approach (for both physician and patient) is to perform rectal disimpaction.

If a high clinical suspicion exists for NBS as the cause of the constipation, is it imperative to avoid treatment with increasing opioids, although clinicians might be initially inclined to do so to control pain. Emergency physicians should consider that patients and providers alike often exhibit frustration due to negative evaluations and increased healthcare utilization by this cohort of patients. Unrecognized NBS can lead to an increase in opioid therapy worsening pain and repeated cycles of treatment failure. Treatment of NBS begins with detoxification, which will likely require hospital admission. Inpatient detoxification involves placing the patient on a continuous opioid drip while the amount is gradually reduced in a structured fashion. The goal of treatment is to achieve dose reductions of 10-33% per day, while providing anxiolytics during withdrawal. Other options for the emergency clinician are nonopioid pain modulators and anxiolytics. Further outpatient support will be necessary for these patients for behavioral modifications, psychological support, and pain management [21, 27, 37]. For opioidinduced constipation, the emergency physician may consider a peripheral acting µ-opioid receptor antagonist (PAMORA) [1].

Treatment of the elderly patient must be individualized based on other comorbidities such as kidney or heart disease. Saline laxatives should be used with caution in this group due to the risk of hypermagnesemia, while nonabsorbable dietary fiber and bulk agents should be avoided in nonambulatory, elderly patients due to the risk for mechanical obstruction. Osmotic laxatives such as polyethylene glycol are a good option for the elderly, but will not likely cause a bowel movement for 2–3 days [1].

In the next section, we will discuss pharmacologic agents in detail.

Pharmacological Agents in the Treatment of Constipation

Bulk-Forming Laxatives

These are often used as first-line agents in patients with chronic constipation and those who cannot take adequate fiber. Examples of bulk-forming laxatives are psyllium (Metamucil), wheat dextrin (Benefiber), methylcellulose (Citrucel), and polycarbophil (Fibercom). The goal of fiber therapy is to trap water in the GI lumen, which in turn stimulates colonic motility and increases frequency of bowel movements. Abdominal bloating is a common adverse effect. These agents are not adequate to treat constipation in the ED and should be considered preventative therapy [1].

Stool Softeners and Lubricants

Examples include mineral oil and docusate sodium. These agents have detergent-like properties that help lubricate the stool by increasing water content and are often tried by the patients prior to ED presentation. They are most likely to work in those with occasional or short duration constipation, and less likely to work in the acute setting or with fecal impaction [1].

Osmotic Laxatives

This class includes saline laxatives such as magnesium citrate and milk of magnesia, as well as poorly absorbed sugars such as lactulose, sorbitol, and polyethylene glycol. Their mechanism of action is to induce water movement into the GI lumen and therefore increase bowel movement frequency by means of softening the stool and increasing volume. Osmotic laxatives are frequently used in the ED as initial treatment for uncomplicated constipation; however, there are instances in which they should be avoided. Due to the potential for electrolyte absorption, saline laxatives are not recommended for patients with renal or cardiac insufficiency. Poorly absorbed sugars, such as lactulose, can lead to abdominal bloating and flatulence, limiting its use due to poor tolerability. Finally, polyethylene glycol is an option, when other therapies have failed if the patient can tolerate oral intake [1].

Stimulant Laxatives

Included in this class are drugs such as bisacodyl, senna, sodium picosulfate, and aloe. These work by increasing intestinal secretions and promoting neurotransmitter release, increasing colonic motility. This class is less helpful in the acute setting but are generally well tolerated with minimal side effects [38].

Enemas

Commonly used enemas include Fleet, soapsuds, and milk and molasses enemas. While these can be very effective in the acute setting, they can also cause severe adverse events (e.g., perforation, metabolic derangements) and as such should be applied carefully [39]. Enemas are contraindicated in those with neutropenia or thrombocytopenia, paralytic ileus, intestinal obstruction, recent colorectal or gynecological surgery, rectal trauma, toxic megacolon, colitis, or recent radiotherapy in the abdominopelvic area [1].

Newer Agents

These include the secretagogues and peripheral acting μ -opioid receptor antagonists (PAMORAs). OIC can be treated with both lubiprostone (a secretagogue) and PAMORAs, such as naloxegol and methylnaltrexone. PAMORAs selectively and competitively bind to the peripheral acting μ -opioid receptors in the enteric nervous system. All PAMORAs have shown effectiveness in treating OIC.

Naloxegol was the first orally dosed PAMORA for the treatment of OIC. It is a PEGylated derivative of naloxone. The recommended dose is 25 mg daily, which may be reduced if patient becomes symptomatic. It is generally well tolerated and will produce a bowel movement in most patients with OIC in approximately 1 week. There are no significant and severe associated adverse events and no reduction in analgesia [1, 25]. It is recommended that other laxatives are discontinued when taking naloxegol and other PAMORAs to decrease adverse side effects such as bloating [40].

Methylnaltrexone is a fast-acting PAMORA, derivative of naltrexone, that is administered subcutaneously and may be useful in the ED [38]. Most patients achieve defecation within 90 minutes of administration. Treatment has not been reported to impact pain control or precipitate opioid withdrawal [1, 41]. Although there is a theoretical risk of perforation among patients with structural abnormalities of the bowel wall, a recent review of 333 adults and children with peritoneal carcinomatosis receiving methylnaltrexone for the treatment of OIC found only one perforation [42].

Naldemedine is an oral PAMORA that has been reported to produce a spontaneous bowel movement in about 16 hours. Naldemedine has no effect on analgesia and there is no increase in opioid withdrawal symptoms [41]. The most effective dose to manage OIC is 0.2 mg [1, 41, 43].

Lubiprostone is a secretagogue that activates type-2 chloride channels in the intestine allowing for an increase in liquid contents and promoting spontaneous bowel movements [25].

Disposition

If the patient is nontoxic appearing and appears to be responding well to the interventions in the ED, they may be discharged with close follow up. However, patients with advanced cancer or unmet palliative care needs may require admission for symptom control or monitoring for complications. Similarly, patients who appear dehydrated, cannot take adequate oral intake, have uncontrolled pain, or constipation that is not resolving must be admitted.

Prevention and Patient Education

Dietary and lifestyle modifications should be encouraged at every visit with the oncologist, as well as discussed during discharge from the ED. Increasing water and fiber content is a simple preventive measure, however, it is a difficult task for some cancer patients due to appetite loss. Probiotics can help as gut microbiota plays a significant role in the GI motility. However, while these recommendations are generally given, the data to support their efficacy are limited [38]. At a minimum, when prescribing opioids for the cancer patient, the clinician should discuss the issue of complication and initiate a bowel regimen.

Conclusion

Constipation in cancer patients offers a specific set of challenges that emergency physicians will need to be familiar with. To improve quality of life in the cancer patients suffering from constipation, more research and data will be needed to further align patient's perception of constipation, clinician's practice, and clinical guidelines. Agreement on a unifying definition can help with earlier recognition and treatment of the disturbing GI symptoms and help reduce morbidity and distress. Questionnaires in existence can be further improved for usability in the ED and to make effective decisions with regard to the best treatment for the chronic constipation patient in the acute setting. Continued education of ED clinicians on the management and opioid-induced constipation and narcotic bowel syndrome is important in the care of cancer patients who often use narcotics. This can help reduce patient distress and increase understanding, which can help the doctor-patient relationship that is often fractured by these patients with difficult clinical presentations and increased healthcare utilization. Increased awareness of NBS by all clinicians can be helpful to these patients as they often need a multidisciplinary team approach including inpatient and outpatient care. In closing, continuing education efforts can greatly help improve the cancer patient's quality of life and help clinicians address the specific set of challenges associated with this cohort.

References

- Larkin PJ, Cherny NI, La Carpia D, Guglielmo M, Ostgathe C, Scotté F, Ripamonti CI. ESMO Guidelines Committee. Diagnosis, assessment and management of constipation in advanced cancer: ESMO clinical practice guidelines. Ann Oncol. 2018;29(Suppl 4):iv111–25.
- Bharucha AE, Pemberton JH, Locke GR. American Gastroenterological Association technical review on constipation. Gastroenterology. 2013;144(1):218–38.

- Laugsand EA, Jakobsen G, Kaasa S, Klepstad P. Inadequate symptom control in advanced cancer patients across Europe. Support Care Cancer. 2011;19(12):2005–14.
- Werth BL, Williams KA, Fisher MJ, Pont LG. Defining constipation to estimate its prevalence in the community: results from a national survey. BMC Gastroenterol. 2019;19(1):75.
- Peppas G, Alexiou VG, Mourtzoukou E, Falagas ME. Epidemiology of constipation in Europe and Oceania: a systematic review. BMC Gastroenterol. 2008;8:5.
- Mugie SM, Benninga MA, Di Lorenzo C. Epidemiology of constipation in children and adults: a systematic review. Best Pract Res Clin Gastroenterol. 2011;25(1):3–18.
- Forootan M, Bagheri N, Darvishi M. Chronic constipation: a review of literature. Medicine (Baltimore). 2018;97(20):e10631.
- Sanchez MI, Bercik P. Epidemiology and burden of chronic constipation. Can J Gastroenterol. 2011;25(Suppl B):11B–5B.
- Larkin PJ, Sykes NP, Centeno C, Ellershaw JE, Elsner F, Eugene B, et al. European Consensus Group on Constipation in Palliative Care. The Management of Constipation in palliative care: clinical practice recommendations. Palliat Med. 2008;22(7):796–807.
- Teunissen SC, de Haes HC, Voest EE, de Graeff A. Does age matter in palliative care? Crit Rev Oncol Hematol. 2006;60(2):152–8.
- Abramowitz L, Béziaud N, Labreze L, Giardina V, Caussé C, Chuberre B, et al. Prevalence and impact of constipation and bowel dysfunction induced by strong opioids: a cross-sectional survey of 520 patients with cancer pain: DYONISOS study. J Med Econ. 2013;16(12):1423–33.
- Sommers T, Corban C, Sengupta N, Jones M, Cheng V, Bollom A, et al. Emergency department burden of constipation in the United States from 2006 to 2011. Am J Gastroenterol. 2015;110(4):572–9.
- The Lancet Gastroenterology Hepatology. The cost of constipation. Lancet Gastroenterol Hepatol. 2019;4(11):811.
- Han D, Iragorri N, Clement F, Lorenzetti D, Spackman E. Cost effectiveness of treatments for chronic constipation: a systematic review. PharmacoEconomics. 2018;36(4):435–49.
- Bowel Interest Group. Cost of Constipation Report, 3rd ed. 2020.
 Aug 2020. https://bowelinterestgroup.co.uk/resources/cost-ofconstipation-report-2020/. Accessed 17 Sep 2020.
- Drossman DA. Functional gastrointestinal disorders: history, pathophysiology, clinical features and Rome IV. Gastroenterology. 2016:S0016-5085(16)00223-7. https://doi.org/10.1053/j.gastro. 2016.02.032.
- Dimidi E, Cox C, Grant R, Scott SM, Whelan K. Perceptions of constipation among the general public and people with constipation differ strikingly from those of general and specialist doctors and the Rome IV criteria. Am J Gastroenterol. 2019;114(7):1116–29.
- Mearin F, Lacy BE, Chang L, Chey WD, Lembo AJ, Simren M, Spiller R. Bowel disorders. Gastroenterology. 2016:S0016-5085(16)00222-5. https://doi.org/10.1053/j.gastro.2016.02.031.
- Simren M, Palsson OS, Whitehead WE. Update on Rome IV criteria for colorectal disorders: implications for clinical practice. Curr Gastroenterol Rep. 2017;19(4):15.
- Andrews CN, Storr M. The pathophysiology of chronic constipation. Can J Gastroenterol. 2011;25(Suppl B):16B–21B.
- Drossman D, Szigethy E. The narcotic bowel syndrome: a recent update. Am J Gastroenterol Suppl. 2014;2(1):22–30.
- McQuade RM, Stojanovska V, Abalo R, Bornstein JC, Nurgali K. Chemotherapy-induced constipation and diarrhea: pathophysiology, current and emerging treatments. Front Pharmacol. 2016;7:414.
- Farmer AD, Holt CB, Downes TJ, Ruggeri E, Del Vecchio S, De Giorgio R. Pathophysiology, diagnosis, and management of opioid-induced constipation. Lancet Gastroenterol Hepatol. 2018;3(3):203–12.

- 24. Hanson B, Siddique SM, Scarlett Y, Sultan S. American Gastroenterological Association Institute Clinical Guidelines Committee. American Gastroenterological Association Institute technical review on the medical management of opioid-induced constipation. Gastroenterology. 2019;156(1):229–253.e5.
- Mesía R, Virizuela Echaburu JA, Gómez J, Sauri T, Serrano G, Pujol E. Opioid-induced constipation in oncological patients: new strategies of management. Curr Treat Options in Oncol. 2019;20(12):91.
- 26. Caterino JM, Adler D, Durham DD, Yeung SJ, Hudson MF, Bastani A, et al. Analysis of diagnoses, symptoms, medications, and admissions among patients with cancer presenting to emergency departments. JAMA Netw Open. 2019;2(3):e190979.
- Farmer AD, Gallagher J, Bruckner-Holt C, Aziz Q. Narcotic bowel syndrome. Lancet Gastroenterol Hepatol. 2017;2(5):361–8.
- Qi Z, Middleton JW, Malcolm A. Bowel dysfunction in spinal cord injury. Curr Gastroenterol Rep. 2018;20(10):47.
- 29. Rentz AM, van Hanswijck de Jonge P, Leyendecker P, Hopp M. Observational, nonintervention, multicenter study for validation of the Bowel Function Index for constipation in European countries. Curr Med Res Opin. 2011;27(1):35–44.
- Long B, Robertson J, Koyfman A. Emergency medicine evaluation and management of small bowel obstruction: evidence-based recommendations. J Emerg Med. 2019;56(2):166–76.
- Kothari K, Friedman B, Grimaldi GM, Hines JJ. Nontraumatic large bowel perforation: spectrum of etiologies and CT findings. Abdom Radiol (NY). 2017;42(11):2597–608.
- 32. Boussios S, Cooke D, Hayward C, Kanellos FS, Tsiouris AK, Chatziantoniou AA, et al. Metastatic spinal cord compression: unraveling the diagnostic and therapeutic challenges. Anticancer Res. 2018;38(9):4987–97.
- Robson P. Metastatic spinal cord compression: a rare but important complication of cancer. Clin Med (Lond). 2014;14(5):542–5.
- Garber JR, Cobin RH, Gharib H, Hennessey JV, Klein I, Mechanick JI, et al. American Association of Clinical Endocrinologists

and American Thyroid Association Taskforce on hypothyroidism in adults. Clinical practice guidelines for hypothyroidism in adults: cosponsored by the American Association of Clinical Endocrinologists and the American Thyroid Association. Endocr Pract. 2012;18(6):988–1028.Erratum in: Endocr Pract. 2013;19(1):175.

- Goldner W. Cancer-related hypercalcemia. J Oncol Pract. 2016;12(5):426–32.
- American College of Gastroenterology Chronic Constipation Task Force. An evidence-based approach to the management of chronic constipation in North America. Am J Gastroenterol. 2005;100(Suppl 1):S1–4.
- Szigethy E, Schwartz M, Drossman D. Narcotic bowel syndrome and opioid-induced constipation. Curr Gastroenterol Rep. 2014;16(10):410.
- Sbahi H, Cash BD. Chronic constipation: a review of current literature. Curr Gastroenterol Rep. 2015;17(12):47.
- 39. Niv G, Grinberg T, Dickman R, Wasserberg N, Niv Y. Perforation and mortality after cleansing enema for acute constipation are not rare but are preventable. Int J Gen Med. 2013;6:323–8.
- 40. Rossi M, Casale G, Badiali D, Aielli F, Aloe Spiriti MA, et al. Opioid-induced bowel dysfunction: suggestions from a multidisciplinary expert board. Support Care Cancer. 2019;27(11):4083–90.
- 41. Candy B, Jones L, Vickerstaff V, Larkin PJ, Stone P. Mu-opioid antagonists for opioid-induced bowel dysfunction in people with cancer and people receiving palliative care. Cochrane Database Syst Rev. 2018;6(6):CD006332.
- Nelson KK, Schattner MA, Mendelsohn RB. Methylnaltrexone is safe in cancer patients with peritoneal carcinomatosis. Sci Rep. 2019;9(1):9625.
- 43. Katakami N, Harada T, Murata T, Shinozaki K, Tsutsumi M, Yokota T, et al. Randomized phase III and extension studies of naldemedine in patients with opioid-induced constipation and cancer. J Clin Oncol. 2017;35(34):3859–66.

Sai-Ching Jim Yeung

Introduction

Routine clinical chemistry analyses including electrolyte and metabolic panels are frequently ordered as part of an emergency department (ED) evaluation to identify abnormalities in electrolyte or acid-base imbalance and to monitor treatments of known abnormalities. In cancer survivors and patients with active malignancies presenting for emergency care, abnormalities in the electrolyte and metabolic panel are very common. Moreover, while serum magnesium is very often not part of the electrolyte and metabolic panel in many EDs, it should be routinely checked in cancer patients because hypomagnesemia is highly prevalent in this population.

In addition to "reflex" treatments in the ED to normalize electrolytes and glucose, early diagnosis or initiation of the diagnostic work-up to identify the underlying cause of the abnormalities will minimize morbidity and mortality as well reduce costs of care. In general, the past medical, medication, and dietary histories will help to determine the cause of abnormalities. A synthesis of physical examination findings with the histories will give clues to specific clinical syndromes.

Case Study

A 65-year-old male previous heavy smoker with chronic obstructive pulmonary disease was recently diagnosed with stage 4 squamous cell lung carcinoma. He had received his

S.-C. J. Yeung (🖂)

third cycle of pembrolizumab and carboplatin plus paclitaxel 11 days ago. He presents with the complaint of palpitations that woke him up this morning. He felt that his heart was racing and would not slow down with rest. He was brought to the ED by ambulance. On arrival, his heart rate was 160/min. His systolic blood pressure was 100 mmHg. He was otherwise at his baseline condition. Stat cardiac monitoring and electrocardiogram showed atrial fibrillation with fast ventricular response. Metoprolol 10 mg was given stat intravenously and was repeated after 15 minutes. The heart rate was controlled at about 100/min. The heart rhythm remained atrial fibrillation. Serum chemistry showed that potassium was 3.3 mEq/L and serum magnesium was 1.4 mg/dL. Potassium chloride and magnesium sulfate infusions were initiated to correct electrolyte abnormalities. Thyroid function tests were sent to the lab, and later TSH was reported to be <0.001 mU/L, with a free thyroxine of 3.0 ng/dL. Cardiology was consulted and evaluation did not reveal underlying atherosclerotic heart disease. Elective cardioversion was performed, and he was successfully converted back to sinus rhythm.

His thyrotoxicosis was due to anti-PD1 antibody-induced thyroiditis and managed with beta-adrenergic blockade. The patient required close follow-up as the thyrotoxic state could resolve, and metoprolol might need to be discontinued. The clinical course of immune thyroiditis may result in hypothyroidism after the thyrotoxic phase, and the patient would then need thyroid hormone replacement.

Metabolic Emergencies

Hypernatremia

Hypernatremia results from loss of balance between sodium and water: excess sodium intake, low water intake, high water loss, and changes in renal reabsorption of water and/ or sodium.



Endocrinology and Metabolism

The original version of this chapter was revised. A correction to this chapter can be found at https://doi.org/10.1007/978-3-030-67123-5_72

Department of Emergency Medicine, Division of Internal Medicine, The University of Texas MD Anderson Cancer Center, Houston, TX, USA e-mail: syeung@mdanderson.org

Causes:

- Inadequate water intake: obstruction of the gastrointestinal tract, treatment-induced nausea and vomiting, and treatment-induced mucositis. A debilitated bed-bound cancer patient may not have free access to water. Primary hypodipsia is loss of thirst due to abnormality in the supraoptic nucleus of the hypothalamus caused by a primary or metastatic malignancy or treatment (surgery or radiation) of a central nervous system tumor.
- Increased water loss: diuretic use, high fever, burns, or diarrhea.
- Iatrogenic causes: inappropriate intravenous (IV) fluid administration, total parenteral nutrition, and dialysis.
- Diabetes insipidus:
 - 1. Central: Caused by changes affecting the anterior pituitary or related hypothalamic nuclei (e.g., neurosurgery, destruction by tumors, hemorrhage, head injury, infarction, and infection).
 - 2. Nephrogenic: Most familial nephrogenic diabetes insipidus cases are caused by mutations of the V2 receptor mutations or aquaporin-2 water channel. However, these are rare among cancer patients. Acquired nephrogenic diabetes insipidus can result from some common drugs (e.g., demeclocycline, lithium, foscarnet, clozapine, amphotericin, glyburide, colchicine, acetohexamide, tolazamide, and methoxyflurane) and chemotherapy agents (e.g., ifosfamide, vinblastine, and streptozocin).

Symptoms The clinical manifestations of hypernatremia are primarily related to cellular dehydration leading to central nervous system dysfunction and are more pronounced with a high level or a fast rate of increase. Thirst is frequently the first symptom. Muscle weakness and central nervous system changes (restlessness, weakness, and lethargy) are usually not manifested until the sodium level is >160 mEq/L, and at this point, the patient may become comatose. Diabetes insipidus is characterized by polyuria, urine hypo-osmolality, and polydipsia. Symptoms of intravascular volume depletion may appear if water loss exceeds water intake.

Diagnosis The cause of hypernatremia is usually evident by history alone. Accurate measurement of fluid intake and output is helpful. A water deprivation test may differentiate between central and nephrogenic diabetes insipidus. Urine osmolality and urinary sodium concentrations should be measured. A serum uric acid >5 mg/dL with polyuria and polydipsia is suggestive of central diabetes insipidus.

Management:

Total body water deficit can be estimated as 0.6 × (body weight in kg) × ((serum sodium level/140) - 1). In patients with acute hypernatremia, free water can be replaced rapidly. In patients with chronic hypernatremia,

the serum sodium level should be decreased by <2 mEq/L/ hour until the symptoms resolve. The remaining water deficit can be corrected in 48 hours.

- Give water enterally, or infuse IV solutions low in sodium (e.g., 0.2% NaCl or dextrose 5% in water).
- Central diabetes insipidus usually is treated with various dosage forms of desmopressin (DDAVP) (5–20 μg intranasally every 12 hours, 1–2 μg subcutaneously once a day, or 0.1–0.2 mg orally twice a day).
- Nephrogenic diabetes insipidus may be managed with a low-salt diet and thiazide diuretics to induce natriuresis and/or indomethacin. Drugs that contribute to nephrogenic diabetes insipidus should be discontinued if possible.

Hyponatremia

Multiple (integumentary, gastrointestinal, cardiovascular, renal, and nervous) organ systems are integrated into regulatory networks for homeostatic control of intravascular volume and serum osmolality, and perturbations of these regulatory networks in patients by cancer or cancer treatments frequently cause hyponatremia.

- Risk factors for hyponatremia include chemotherapy, nausea and vomiting, hydration with hypotonic fluid, pain, opioids, and stress
- Hypothyroidism (see section about thyroid dysfunction below)
- Adrenal insufficiency (see section about adrenal dysfunction below)
- Congestive heart failure
- Cirrhosis
 - Syndrome of inappropriate secretion of antidiuretic hormone (SIADH), characterized by normal or increased intravascular volume, low serum osmolality, and inappropriately high urine osmolality in the absence of diuretics, cirrhosis, heart failure, hypothyroidism, and adrenal insufficiency:
 - 1. Cancers secreting vasopressin (e.g., about 15% of SCLC, 1% of other lung cancers, and 3% of squamous cell head and neck cancers) [1]
 - 2. Abnormal secretory stimuli for ADH (e.g., intrathoracic infection, positive pressure ventilation)
 - 3. Cytotoxic chemotherapeutic agents affecting paraventricular and supraoptic neurons (vinca alkaloid [vincristine and vinblastine], high-dose cyclophosphamide)
- Renal salt wasting:
 - 1. Tumor-induced: Mediated by atrial natriuretic peptide [2]
 - Drug-induced: Damage to the renal tubules and resulting defects in salt and water transport may be the major cause of hyponatremia associated with low-dose cyclophosphamide therapy [3] and platinum compounds [4]

Symptoms The signs and symptoms of hyponatremia include general weakness, fatigue, nausea, and vomiting, all of which are nonspecific. The severity of symptoms depends on the rate of decline and degree of hypo-osmolality. Significant symptoms usually appear as the serum sodium concentration falls below 120 mEq/L. Neurologic symptoms of headache, behavioral changes, lethargy, confusion, seizure, stupor, and coma may manifest, while progressive cerebral edema causes brain damage, brain stem herniation, respiratory failure, and death.

Identifying the causes of hyponatremia requires additional laboratory evaluations, including urinary sodium measurement, thyroid and adrenal function tests, and correlation with clinical history.

Diagnosis Figure 29.1 outlines the evaluation and treatment of hyponatremia. Evaluation of intravascular volume status is very important in diagnosing the underlying cause of hyponatremia. Hypotonicity must be confirmed by measuring osmolality. Pseudohyponatremia caused by hyperlipidemia, hyperproteinemia, severe hyperglycemia, and administration of hypertonic mannitol should be excluded.

Management:

- If the patient is not hypovolemic, free water intake may be restricted to 500–800 mL/day.
- There are several ways to increase free water excretion:
- 1. Drug-induced nephrogenic diabetes insipidus (e.g., demeclocycline 600–1200 mg/day)
- 2. Loop diuretics (e.g., furosemide 20-40 mg/day)
- Blockade of V2 receptors to promote free water excretion (aquaresis) (e.g., conivaptan, lixivaptan, tolvaptan, and satavaptan) [5]
- Fludrocortisone (0.1–0.6 mg/day) is a mineralocorticoid that may be used to decrease renal sodium excretion.
- For patients with hypovolemia, oral intake of sodium may be increased by sodium chloride tablets along with intravascular volume expansion by infusing normal saline (0.9% NaCl).
- For emergent cases of hyponatremia in which central nervous system symptoms are evident and significant, infusion of hypertonic saline (3% NaCl) at a rate of 1 mL/kg/ hour may be indicated with close monitoring (in an intensive care unit).



Fig. 29.1 Algorithm for the evaluation and management of hyponatremia

Hyperkalemia

In the absence of excessive intake of potassium, hyperkalemia is an electrolyte abnormality often associated with renal abnormalities in cancer patients.

Causes:

- Diminished renal excretion of potassium occurs in patients with acute or chronic renal failure, renal hypoperfusion, or type 4 renal tubular acidosis.
- Drugs that can lead to decreased potassium excretion include potassium-sparing diuretics and angiotensin-converting enzyme inhibitors.
- Excessive oral potassium supplementation and inappropriate potassium content in IV fluid or total parenteral nutrition.
- A significant release of intracellular potassium will cause hyperkalemia, as in the case of tumor lysis syndrome.
- Transcellular shifts of potassium may be seen in insulin deficiency, β-blocker therapy, and acidemia, elevating serum potassium levels.
- Drug-induced hyperkalemia often occurs with preexisting impaired renal excretion of potassium. The drugs commonly used by cancer patients that cause hyperkalemia include cyclosporin A, tacrolimus, heparin, mitomycin-C, and pentamidine.

Symptoms Severe clinical manifestations of hyperkalemia are usually absent until the serum level is >7.5 mEq/L. Some

patients (e.g., those with chronic renal failure) can tolerate high serum potassium levels without having any clinical signs or symptoms. Hyperkalemia causes depolarization of excitable membranes. This membrane depolarization leads to the excitability of nerves and muscles, causing cramps, muscle weakness, and paralysis. At >7.5 mEq/L, nonspecific symptoms, such as muscle weakness, cramping and paralysis of different muscle groups, may occur.

Diagnosis The most vital organ with excitable membranes is the heart. Electrocardiogram (EKG) changes and potentially fatal arrhythmias may be present. An early EKG abnormality associated with hyperkalemia is peak T waves (Fig. 29.2) followed by progressive QRS widening to a "sinusoidal" wave (Fig. 29.3) [6]. Ventricular tachycardia, fibrillation, and asystole may occur.

Hypotension and hypoglycemia with hyperkalemia suggest possible adrenal insufficiency. Serum electrolytes, blood urea nitrogen, serum creatinine, urinalysis, urine electrolytes, and arterial blood gases will often elucidate the cause of hyperkalemia.

Management:

 If possible, discontinue medications that may contribute to hyperkalemia (e.g., potassium supplements, spironolactone, amiloride, β-adrenergic blockers, nonsteroidal anti-inflammatory drugs, angiotensin receptor blockers, and angiotensin-converting enzyme inhibitors).



Fig. 29.2 Peaked T wave in an electrocardiogram of a hyperkalemic patient. The serum potassium was 7.3 mEq/L



Fig. 29.3 EKG changes in the presence of electrolyte abnormalities [6]

- Hyperkalemia due to calcineurin inhibitors cyclosporine and tacrolimus may respond to treatment with fludrocortisone.
- For severe hyperkalemia (>6.5 mEq/L) and/or EKG changes, monitor EKG continuously and treat with the following:
 - 1. IV calcium (1–2 g of calcium gluconate or 0.5–1.0 g of chloride).
 - 2. IV sodium bicarbonate (1 mEq/kg).
 - 3. IV glucose (usually 25 g) plus 6–8 U of regular insulin.
 - 4. β-adrenergic agonists (e.g., albuterol 2.5 mg nebulized inhalation). Loop diuretics may be used to promote natriuresis and kaliuresis.
- Ion exchange resins, such as sodium polystyrene sulfonate (Kayexalate), which can be administered orally (15– 30 g/dose) or rectally (30–60 g/dose) as a retention enema,

may remove potassium from the body via the gastrointestinal tract.

• Emergent hemodialysis may be used in refractory cases.

Hypokalemia

Hypokalemia is a very common electrolyte abnormality in cancer patients.

- Potassium intake in cancer patients may decrease for various reasons, such as nausea, vomiting, anorexia and gastrointestinal obstruction.
- Potassium may be lost from the gastrointestinal tract via vomiting or diarrhea, and from the kidneys as a result of intrinsic tubular defects and type 1 renal tubular acidosis.

- Drug-related effects (e.g., loop diuretics, aminoglycosides, cyclophosphamide, ifosfamide, carboplatin, cisplatin, and amphotericin B).
- Hypokalemia owing to excess mineralocorticoid activity may result from deregulated aldosterone production by adrenal tumors or other renin-secretion cancers [renal (Wilms tumor, renal cell carcinoma, or hemangiopericytoma), lung (SCLC, adenocarcinoma), hepatic, pancreatic, or ovarian carcinomas] [7].
- Exogenous corticosteroids, fludrocortisone, and Cushing syndrome, which includes ectopic secretion of ACTH by some cancers (e.g., small cell lung cancer [SCLC], carcinoid tumors).
- Alkalosis, either respiratory or, on a larger scale, metabolic, may precipitate hypokalemia via a transcellular potassium shift. Drugs that cause potassium redistribution include insulin, β-adrenergic agonists, theophylline, and chloroquine.

Symptoms Patients with mild hypokalemia (3.0–3.5 mEq/L) are usually asymptomatic. In those with severe hypokalemia (<3.0 mEq/L), symptoms may range from mild to severe (and potentially fatal). Cardiac manifestations range from flat T waves, T-wave depression, and prominent U waves to serious arrhythmias (ventricular tachycardia or fibrillation) (see Fig. 29.3). Neurologic manifestations include muscle weakness, paresthesia, and paralysis.

Diagnosis Medications and dietary histories will help determine the cause of hypokalemia. Physical examination will give clues regarding Cushing syndrome. Measurement of serum electrolytes, including magnesium, blood urea nitrogen, and creatinine; urinalysis; and urine electrolyte measurement will help to diagnose renal potassium loss.

Management The oral route for potassium replacement is preferred over other routes if feasible. The IV route may be used in patients with severe hypokalemia or those unable to tolerate enteral replacement. The rate of IV administration should not exceed 20 mEq/hour diluted in IV fluid through a peripheral vein. The infusion rate may be as high as 40 mEq/hour through a central venous catheter. In general, the relationship between the degree of hypokalemia and total body deficit is linear. For each 1 mEq/L decrease in serum potassium level, the total body deficit would be about 300 mEq. This total body deficit may be corrected over days. Almost half of the cancer patients with hypokalemia have concurrent hypomagnesemia. Potassium-sparing diuretics, such as amiloride and spironolactone, inhibit potassium excretion and may have a role in decreasing renal potassium wasting.

Hypermagnesemia

Hypermagnesemia is uncommon.

Causes:

- Renal failure.
- Increased intake of magnesium in the presence of renal insufficiency.
- Excessive magnesium in IV fluid or parenteral nutrition.
- In the absence of renal insufficiency, hypermagnesemia owing to excessive intake of magnesium is very rare, as excess magnesium in the gastrointestinal tract leads to diarrhea.

Symptoms The clinical manifestations of hypermagnesemia correlate well with the serum level of magnesium. Early signs include nausea, vomiting, weakness, and cutaneous flushing, which can occur when the serum magnesium level is greater than 3 mg/dL. With levels greater than 4 mg/dL, hyporeflexia and loss of deep tendon reflexes may occur. At levels greater than 5 mg/dL, hypotension and EKG changes (QRS widening, QT and PR prolongation, and conduction abnormalities) may occur (see Fig. 29.3). Respiratory depression, coma, and complete heart block may occur at levels greater than 9 mg/dL. Asystole and cardiac arrest occurs at levels greater than 10 mg/dL.

Diagnosis Excessive magnesium intake usually is evident in the patient's dietary and medication histories. Renal function should be assessed by measuring blood urea nitrogen and serum creatinine.

Management:

- Medications and IV fluids containing magnesium should be discontinued.
- Patients with mild symptoms and normal renal function can be observed without intervention.
- Magnesium excretion can be accelerated by hydration with crystalloid fluid and loop diuretics.
- In severe cases with hypotension and/or cardiac arrhythmia, IV calcium should be administered to reverse respiratory depression, hypotension, and cardiac arrhythmia. Emergent dialysis should be considered to correct life-threatening hypermagnesemia in the presence of impaired renal function.

Hypomagnesemia

Magnesium is a major inorganic cation in the body, of which only 1-2% is present in the extracellular space.

Hypomagnesemia is defined as a plasma serum concentration of magnesium less than 1.5 mg/dL. However, magnesium levels that are persistently less than 1.8 mg/dL indicate depletion of total body magnesium. The prevalence of hypomagnesemia in hospitalized cancer patients is about 20%.

Causes:

- Low oral intake, impairment of renal reabsorption, prolonged IV feeding, chronic alcoholism, intestinal malabsorption, and diarrhea.
- The renal toxicity of chemotherapy (e.g., platinum-based drugs, cyclophosphamide, ifosfamide) or anti-infective medications (e.g., amphotericin, aminoglycosides) will also cause hypomagnesemia. Platinum agents cause morphologically evident nephrotoxicity that may result in long-term or even permanent damage to renal functions. Hypomagnesemia occurs in approximately 90% of patients treated with cisplatin [8], and 10% of the hypomagnesemic patients have symptoms of muscle weakness, tremors, and dizziness. Hypomagnesemia may persist long after cessation of cisplatin therapy.
- Anti-EGFR targeted therapy (cetuximab, panitumumab, and anti-EGFR small molecular inhibitors such as gefitinib) reduces renal magnesium reabsorption through a reversible effect of epidermal growth factor receptor (EGFR) inhibition in the renal distal convoluted tubule without inflicting extensive cell or organ injury [9].

Symptoms Magnesium is needed for a wide variety of enzymatic reactions, including those involving ATP and nucleic acid metabolism. Magnesium is also directly involved in the regulation of calcium and potassium metabolism. The clinical manifestations of hypomagnesemia may be nonspecific and include anorexia, nausea, vomiting, lethargy, dizziness, muscle weakness, tremor, muscle fasciculation, tetany, and tonic-clonic seizures.

Diagnosis Hypomagnesemia is often associated with other electrolyte abnormalities, such as hypokalemia and hypocalcemia [10]. Concurrent measurement of other electrolytes, such as calcium, phosphate, and potassium, should be considered. Significant hypomagnesemia is associated with EKG changes (see Fig. 29.3). According to CTCAE version 4, hypomagnesemia is Grade 1 when magnesium is below normal to ≥ 0.5 mM; Grade 2, <0.5–0.4 mM; Grade 3, <0.4– 0.3 mM; and Grade 4, <0.3 mM with life-threatening consequences.

Management Magnesium replacement is indicated in cancer patients when serum magnesium is repeatedly below normal. Oral replacement is preferred over parenteral when feasible. However, diarrhea may be a dose-limiting side effect. When IV replacement is required, the usual practice is to replace half of the estimated dose over 1 day and the remaining half over the next 3–4 days. Hypomagnesemia induced by anti-EGFR-therapy is reversible.

Hypercalcemia

The incidence of hypercalcemia in cancer patients is about 1% [11].

Causes:

- Hypercalcemia of malignancy accounts for more than 90% of hypercalcemia cases:
 - Parathyroid hormone-related protein (PTHrP)mediated hypercalcemia [12, 13] is a paraneoplastic syndrome associated with a short survival. PTHrP causes hypercalcemia by binding to the parathyroid hormone (PTH) receptor and activating the expression of an osteoblast-specific cell surface protein, RANK ligand (RANKL). Interaction between RANKL and the RANK receptor on the osteoclast precursor causes increased osteoclast differentiation, bone resorption, and hypercalcemia. PTHrP production is found commonly in squamous cell carcinoma, breast, neuroendocrine, renal, and prostate cancers, as well as melanoma.
 - 2. Other tumor-secreted humoral factors, such as interleukin-1 and -6, prostaglandins and tumor necrosis factor, may contribute to hypercalcemia.
- 3. In multiple myeloma, increased expression of RANKL causing localized osteoclast proliferation appears to be the most important cause [13].
- Lymphomas commonly express 1α-hydroxylase, the enzyme that converts 25-hydroxy vitamin D3 to 1,25-dihydroxy vitamin D3 (calcitriol), leading to increased gastrointestinal absorption of calcium [14].
- Primary hyperparathyroidism should be considered in the differential diagnosis of hypercalcemia. No cancer treatment has been identified as a cause of hypercalcemia, except that low-dose (2–7.5 Gy) external-beam irradiation of the head and neck area increases the incidence of primary hyperparathyroidism by 2.5- to 3-fold many years after irradiation of the neck (29–47 years) [15]. Primary hyperparathyroidism may also develop in multiple endocrine neoplasia (MEN), especially type 1.

Symptoms Patients with mild hypercalcemia (calcium level <12 mg/dL) usually have no symptoms, whereas those with moderate or severe hypercalcemia are frequently symp-

tomatic. Central nervous system symptoms are lethargy, ataxia, stupor, coma, mental status changes, and psychosis. Gastrointestinal tract symptoms are anorexia, nausea, constipation, ileus, dyspepsia, and pancreatitis. Renal signs are polyuria, nephrolithiasis, and nephrocalcinosis. Cardiovascular manifestations can be a short QT interval, ST segment depression, sinus arrest, and atrioventricular block (see Fig. 29.3). Musculoskeletal symptoms are myalgia, arthralgia, and weakness. Severe hypercalcemia (>13 mg/dL) frequently causes depression of cerebral function or coma.

Diagnosis Serum calcium levels should be interpreted in the context of protein binding (corrected calcium level = $[0.8 \times (normal albumin level - patient's albumin level)]$ + serum calcium level). Measurement of the ionized calcium can confirm hypercalcemia. Laboratory studies of the following help diagnose the etiology of hypercalcemia: intact PTH, PTH-related protein, 25-hydroxy vitamin D₃, and 1,25-dihydroxy vitamin D₃. The combination of hypercalcemia and an elevated PTH level combined with increased urinary calcium excretion provides reasonable evidence for primary hyperparathyroidism. Suppression of the PTH below the normal range is found in PTHrP or calcitriol-mediated hypercalcemia, which can be diagnosed by measuring 25-hydroxy vitamin D3 and 1,25-dihydroxy vitamin D3.

PTHrP-mediated hypercalcemia is characterized by a suppressed PTH level and a low or normal calcitriol level. This contrasts with the finding of elevated PTH and calcitriol levels in primary hyperparathyroidism. The characteristic clinical features of hypercalcemia in lymphoma include a suppressed serum PTH, a normal or slightly increased phosphate level due to suppression of PTH, hypercalciuria, absence of bone metastasis, and an elevated serum calcitriol level [14].

Management:

- The initial and first-line treatment of hypercalcemia is hydration by infusion of normal saline at rates between 100 and 300 mL/h. Hydration alone can lower the serum calcium by 10% or more over 6–12 hours. In patients with overall fluid overload, use of a loop diuretic would be helpful.
- Calcium-containing medications and thiazide diuretics (which inhibit renal tubular excretion of calcium) should be discontinued.
- Bisphosphonates inhibit bone resorption by osteoclasts, and their peak impact on hypercalcemia is usually seen after a couple of days. Zoledronate (4–6 mg IV over 30 minutes) [16] is more widely used than pamidronate (60–90 mg IV over 4–24 hours) because of its higher potency and efficacy [17].

- Second-line agents include calcitonin (salmon calcitonin 4 IU/kg subcutaneously every 12 hours). Calcitonin has a rapid onset of action, but its effectiveness may decrease within 2–3 days.
- Glucocorticoids (40–60 mg/d prednisone equivalent) may be used in hypercalcemia associated with myeloma and lymphoma.
- Denosumab (anti-RANKL antibody) is a new drug for hypercalcemia of malignancy [18].
- Primary hyperparathyroidism can be cured via parathyroidectomy. Removal of adenoma is usually curative, but in the context of MEN1, the surgical procedure of choice is three-and-a-half-gland parathyroidectomy [19].

Hypocalcemia

Hypocalcemia is a common complication of chemotherapy [20].

Causes:

- Nephrotoxicity of platinum compounds: Hypocalcemia has been reported in 6–20% of cisplatin-treated and 16–31% of carboplatin-treated patients. Hypomagnesemia may decrease secretion of PTH and reduce its calcium-mobilizing effects. Hypomagnesemia also inhibits the formation of 1,25-dihydroxy vitamin D₃. Platinum compounds may inhibit the mitochondrial function in the kidneys and thereby inhibit conversion of 25-hydroxy vitamin D₃ to 1,25-dihydroxy vitamin D₃.
- Plicamycin (mithramycin) and dactinomycin are two infrequently used antineoplastic agents that are known to cause hypocalcemia.
- Surgical procedures in the neck that sacrificed or damaged the parathyroid glands (e.g., total laryngectomy, total thyroidectomy) can cause primary hypoparathyroidism, leading to hypocalcemia.
- Vitamin D deficiency causes rickets and osteomalacia along with hypocalcemia.
- Tumor lysis syndrome.

Symptoms Hypocalcemia can be asymptomatic if it is mild. Life-threatening problems such as seizures, cardiac dysrhythmias, and laryngospasm can occur if hypocalcemia is severe. Acute hypocalcemia is characterized by neuromuscular irritability. Acute symptoms are muscle weakness, paresthesia, spasm, tetany, hyperreflexia, Chvostek sign, Trousseau sign, seizure, bronchospasm, laryngeal spasm, and respiratory failure. Cardiovascular presentations are bradycardia, hypotension, QT-interval prolongation (see Fig. 29.3), congestive heart failure, and cardiac arrest. Chronic hypocalcemia with hypoparathyroidism causes extrapyramidal disorders and cataracts, as well as skin and hair changes.

Diagnosis Measuring ionized calcium can exclude pseudohypocalcemia owing to low albumin and serum protein levels. In most cancer patients, the etiology of hypocalcemia is obvious. The major causes of hypocalcemia are hypoparathyroidism, hypomagnesemia, and chemotherapy toxicity. If the cause of hypocalcemia is not clear, laboratory analysis of intact PTH, magnesium, phosphate, 25-hydroxy vitamin D3, 1,25-dihydroxy vitamin D3, creatinine and 24-hour urinary calcium levels is helpful.

Management:

- Severe hypocalcemia is treated parenterally with IV calcium chloride (0.5–1.0 g) or gluconate (1–2 g) over 5–10 minutes.
- Hypomagnesemia is a common cause of hypocalcemia. Concurrent hypomagnesemia should be treated with IV magnesium sulfate followed by oral replacement.
- Chronic hypocalcemia is treated with oral calcium preparations (e.g., gluconate, carbonate) containing 1–2 g of elemental calcium per day.
- Patients with hypoparathyroidism often require life-long supplementation of calcium and vitamin D. Vitamin D supplements can be given in 1-hydroxylated form or as calcitriol.
- Recombinant PTH_{1-34} (teriparatide) is now approved for the treatment of osteoporosis. Its use in hypoparathyroid-ism remains to be studied.
- For hypocalcemia secondary to hyperphosphatemia, hyperphosphatemia often requires correction first. See below for management of hyperphosphatemia.

Hyperphosphatemia

In the absence of renal failure, the fasting serum phosphate level is determined primarily according to the renal tubular reabsorption rate.

Causes:

- A massive amount of phosphate can be released into the extracellular fluid via extensive cellular breakdown (e.g., tumor lysis syndrome, rhabdomyolysis, and hemolysis).
- Translocation of phosphate from cells in response to metabolic or respiratory alkalosis can lead to acute hyperphosphatemia.
- Chronic hyperphosphatemia is present in patients with hypoparathyroidism on long-term treatment with oral calcium and vitamin D.
- Excess phosphate intake (e.g., phosphate-containing laxatives), especially in the presence of renal insufficiency.

Symptoms The clinical manifestations of acute hyperphosphatemia are similar to those of associated hypocalcemia. Paresthesia, muscle cramps, tetany, and QT-interval prolongation may be induced directly by severe hyperphosphatemia. Chronic hyperphosphatemia, especially associated with hypercalcemia, may lead to diffuse visceral deposition of calcium phosphate. Deposition of calcium phosphate in the kidneys may lead to renal failure.

Diagnosis In patients with hyperglobulinemia, pseudohyperphosphatemia must be excluded with a specimen that is free of protein (removed via precipitation with sulfosalicylic acid). In those with hyperphosphatemia, renal function must be assessed. In addition, measurement of lactic dehydrogenase, uric acid, potassium, and calcium levels is necessary to diagnose and manage hyperphosphatemia due to extensive cellular lysis.

Management:

- In patients with normal renal function, infusion of isotonic saline increases phosphate excretion.
- Administration of dextrose and insulin drives phosphate into cells, temporarily lowering the serum phosphate level.
- When hyperphosphatemia is life-threatening, hemodialysis or peritoneal analysis should be considered.
- Blocking phosphorus absorption in the gastrointestinal tract:
 - 1. Aluminum hydroxide
 - 2. Calcium-based phosphate binders (e.g., calcium acetate)
 - Nonabsorbable phosphate binders that are aluminumand calcium-free (sevelamer 800–1600 mg with each meal)

Hypophosphatemia

Hypophosphatemia is found in about 2-3% of all hospitalized patients and about 30% of cancer patients.

- Relative nutritional deficiency:
 - Hypophosphatemia in malnourished patients (especially alcoholics) is due to a combination of magnesium deficiency, vitamin D deficiency, and malabsorption. Acute hypophosphatemia may occur in hospitalized patients with serious illnesses and preexisting phosphate depletion.
 - 2. Refeeding with high-calorie contents in severely malnourished patients.

- 3. Rapid cancer proliferation may cause hypophosphatemia (e.g., Burkitt's lymphoma).
- 4. Rapid normal cell proliferation, as with the use of granulocyte colony-stimulating factors, hematopoietic reconstitution after stem cell transplantation, or stem cell harvesting in preparation for transplantation.
- Renal wasting of phosphorus and calcium:
 - Tumor-induced osteomalacia is a rare paraneoplastic syndrome characterized by hypophosphatemia, excessive urinary phosphate loss, reduced 1,25-dihydroxy vitamin D concentrations, and osteomalacia. Fibroblast growth factor-23 may be the humoral mediator of this paraneoplastic syndrome [21]. Tumors that produce this clinical syndrome include mesenchymal tumors (osteoblastomas, giant cell osteosarcomas, hemangiopericytomas, hemangiomas, nonossifying fibromas) [22] and, rarely, malignant tumors such as prostate or lung cancer.
 - 2. Intrinsic renal tubular defect in phosphate reabsorption, or acquired renal tubular defect (e.g., after ifosfamide [23], cisplatin [24], and estramustine [25]).
- Transcellular shift of phosphate (e.g., respiratory alkalosis, IV glucose administration, hyperalimentation, gramnegative sepsis, or insulin therapy).
- Elevated PTH (primary hyperparathyroidism) or PTHrP (hypercalcemia of malignancy).
- Accelerated bone formation (e.g., extensive blastic bone metastasis in prostate cancer, hungry bone syndrome after resection of parathyroid adenomas).
- Loss of liver function: The liver also plays a significant role in phosphate homeostasis. Serum phosphate decreases after right or extended right hepatic lobectomy and in hepatocellular carcinoma complicating liver cirrhosis.
- Consumption of aluminum-containing medications/ antacids.

Symptoms Acute severe hypophosphatemia may lead to general neurologic findings such as lethargy, confusion, disorientation, hallucinations, and focal neurologic findings such as dysarthria, dysphagia, oculomotor palsy, anisocoria, nystagmus, ataxia, cerebellar tremor, ballismus, hyporeflexia, distal sensory deficits, paresthesia, and hyperesthesia. Severe neurologic symptoms, such as muscle paralysis, seizure, and coma, are observed only when the serum phosphate level is <0.8 mg/dL. In severe hypophosphatemia, reversible left ventricular dysfunction can occur.

Muscle weakness is the most common complaint. Bone pain is another prominent complaint of phosphate-depleted patients. Prolonged hypophosphatemia leads to rickets. Osteomalacia, a condition characterized by unmineralized bone matrix, should be considered in osteopenic patients with bone pain and proximal myopathy. Waddling gait, bone tenderness, pseudo-fractures, and fractures can occur in patients with chronic hypophosphatemia. Osteomalacia and moderate to severe proximal myopathy are also characteristics of tumor-induced osteomalacia [26].

Diagnosis Measurement of renal function and potassium, magnesium, ionized calcium, vitamin D metabolites, and PTH level is helpful in the initial evaluation of the cause of hypophosphatemia. If urinary loss of phosphate is suspected, urine should be collected to measure the renal phosphate threshold/glomerular filtration rate to confirm phosphaturia.

Management:

- Significant hypophosphatemia (phosphate level less than 2 mg/dL), especially in the context of underlying phosphate depletion, should be corrected promptly.
- Phosphate can be safely administered IV at an initial dose of 0.2–0.8 mmol/kg over 6 hours (i.e., 10–50 mmol over 6 hours). Higher doses (1.5–3.0 mmol/kg over 12 hours) should be reserved for patients with phosphate levels less than 1.5 mg/dL and normal renal function.
- Mild hypophosphatemia can be treated with oral phosphate in divided doses of 750–2000 mg/day.
- For tumor-induced osteomalacia, oral or IV supplementation of phosphate combined with vitamin D therapy is generally effective for eradicating or improving clinical symptoms. Complete surgical removal of the tumor is generally curative.

Hyperglycemia

Diabetes mellitus type 2 (DM2) is a common disease, and a large number of cancer patients have co-existing DM2. Extensive epidemiologic data suggest an important role of DM2 in carcinogenesis [27–32] and cancer survival [33], and DM2 is associated with an elevated risk of pancreatic, liver, colon, gastric, breast, and endometrial cancer [27–32].

- The administration of glucocorticoids (e.g., for antineoplastic therapy in combination regimens, for edema of brain metastasis, for prevention of transplant rejection, for graft-versus-host disease in BMT, and for nausea/ vomiting) is probably the most common cause of druginduced diabetes in cancer patients.
- Treatment with streptozocin [34] or L-asparaginase [35] may result in insulin-deficient diabetes mellitus.
- Diabetes mellitus may also develop as a consequence of serious pancreatitis secondary to treatment with L-asparaginase.
- Interleukin-2 and interferons may cause toxicity to pancreatic β cells and lead to insulin-dependent diabetes [36].

Immune checkpoint inhibitors may cause acute development of type I diabetes and precipitate the serious complication of diabetic ketoacidosis.

- Some targeted therapy antineoplastic agents interfere with the insulin signaling pathway and can cause hyper-glycemia. PI3K inhibitors (alpelisib, copanlisib, duvelisib, and idelalisib) and mTOR inhibitors (e.g., rapamycin (sirolimus), everolimus, and temsirolimus) are associated with a high incidence of hyperglycemia [37, 38]. Some tyrosine kinase inhibitors (TKIs) (e.g., niltinib and sunitinib) are associated with hyperglycemia, but some TKIs (e.g., imatinib and pazopanib) may be associated either with hyperglycemia or with hyporglycemia [38].
- Tacrolimus, an immunosuppressive agent used to prevent graft-versus-host disease in bone marrow stem cell transplantation, also increases the incidence of diabetes, perhaps by damaging pancreatic β cells [39]. Patients who received allogenic stem cell transplantation are likely to be receiving both glucocorticoids, cyclosporine A and tacrolimus, and are particularly at risk for developing diabetes mellitus [40].

Symptoms Most patients with significant hyperglycemia have symptoms of polydipsia, polyuria, and polyphagia. Dehydration of the lenses owing to hyperglycemia leads to blurry vision. Patients with hyperosmolar nonketotic coma experience mental status changes, hypotension, and severe dehydration. Nausea, vomiting, and abdominal pain are present in almost half of patients with diabetic ketoacidosis. Tachypnea with Kussmaul respiration, tachycardia, hypotension, orthostatic blood pressure changes, acetone breaths, and severe signs of dehydration can be present in patients with diabetic ketoacidosis.

Diagnosis A random plasma glucose >200 mg/dL or a fasting plasma glucose >126 mg/dL on more than one occasion can indicate diabetes mellitus. Abnormal glucose may require further diagnostic evaluation with a glucose tolerance test or mixed meal tolerance test or glycosylated hemoglobin (hemoglobin A1C).

Diabetic ketoacidosis is diagnosed according to the triad of metabolic acidosis, hyperglycemia, and the presence of ketone bodies in the urine or blood. Arterial blood gas testing will demonstrate acidemia and respiratory compensation for metabolic acidosis by hyperventilation. Also, the anion gap will be elevated, and serum ketone testing will be positive. A urine dipstick test for ketones can provide timely information for a quick bedside diagnosis. Absence of ketones from the urine practically excludes diabetic ketoacidosis. Leukocytosis may be associated with ketosis, but an infection must be considered as a precipitating factor for diabetic ketoacidosis. The serum creatinine level can be falsely elevated because of ketosis. Potassium, phosphate, and magnesium abnormalities result from transcellular shifts caused by acidosis.

In hyperosmolar hyperglycemic nonketotic coma, the plasma glucose level may be >800 mg/dL and the serum osmolality may be >100 mOsm above normal. Mild ketosis may be present because of starvation, but ketoacidosis is absent. In severe cases, when volume depletion compromises tissue perfusion, lactic acidosis will develop.

In immunocompromised cancer patients in particular, sepsis must be ruled out as the precipitating event for diabetic ketoacidosis or hyperosmolar hyperglycemic coma.

Management:

- In general, insulin will be needed in patients who are insulin deficient.
- Diabetic ketoacidosis is decompensated catabolism triggered by a relative or absolute deficiency in insulin secretion.
- Treatment of diabetic ketoacidosis or hyperosmolar hyperglycemic coma:
 - 1. Hydration with IV crystalloid fluid.
 - Regular insulin, which usually is given as an IV bolus of 0.1 U/kg, followed by a maintenance IV infusion of 0.1 U/kg/hour. The amount of insulin required for treatment of hyperosmolar hyperglycemic coma may be less than that required for diabetic ketoacidosis.
 - 3. Correction of electrolyte abnormalities, but beware of transcellular shift of electrolytes related to blood pH and the effect of insulin.
 - 4. Identification of the precipitating factors (particularly important to rule out sepsis).

Hypoglycemia

Glucagon and epinephrine immediately stimulate hepatic glycogenolysis followed by gluconeogenesis and are the two major counterregulatory hormones in response to hypoglycemia. Other counterregulatory hormones are norepinephrine, cortisol, and growth hormone, but their effects are delayed.

- Cancer-related malnutrition, fat and muscle wasting, cirrhosis, and extensive liver metastases may impair glycogenolysis and gluconeogenesis.
- Adrenal insufficiency is associated with hypoglycemia; refer to the section on adrenal crisis below for detailed discussion.
- For diabetic cancer patients receiving sulfonylurea or insulin, the most common cause of hypoglycemia may be delayed or decreased food intake. The kidneys contribute to overall gluconeogenesis during hypoglycemia stress in

about one third of cases and are important to extrahepatic degradation of insulin. Moreover, a number of oral hypoglycemic drugs are excreted by the kidneys. Therefore, decline in renal function often leads to hypoglycemic episodes in diabetic patients.

- Tumor-induced hypoglycemia is an uncommon but challenging cause of morbidity for cancer patients. Three different clinical syndromes have been identified: (1) secretion of insulin by islet cell malignancy; (2) insufficient gluconeogenesis due to near complete replacement of hepatic parenchyma by tumor; and (3) secretion of insulin-like growth factor II (IGF2), which activates the insulin receptor and causes hypoglycemia [41–43] by tumors (e.g., fibrosarcomas, hemangiopericytomas, and hepatomas).
- Excessive glucose consumption by large tumors also may cause hypoglycemia. Hypoglycemia may also occur in patients with lactic acidosis in the context of end-stage leukemia or lymphoma [44].

Symptoms A pattern of hypoglycemia symptoms progresses as the availability of glucose to the brain decreases. At a plasma glucose level of about 70 mg/dL, brain glucose uptake can be reduced and counterregulatory hormone responses are triggered. At <60 mg/dL, autonomic symptoms, such as hunger, anxiety, palpitations, sweating, and nausea, become evident. When glucose <50 mg/dL, neuroglycopenic symptoms of blurry vision, slurred speech, inability to concentrate, and confusion appear. When glucose <40 mg/dL, the patient may become drowsy, confused, or combative. A prolonged decrease below 30 mg/dL can cause seizures, permanent neurologic damage, and death.

Diagnosis Hypoglycemia is diagnosed by blood chemistry, but rapid bedside measurement of blood glucose should be expeditiously performed in the evaluation of all ED patients with altered mental status. The timing of symptoms relative to a fasting or postprandial state or to anti-diabetic medications can distinguish among various causes of hypoglycemia.

The most common presentation for paraneoplastic syndromes of hypoglycemia is fasting hypoglycemia, and patients are most likely to develop symptoms during normal periods of fasting, particularly during nocturnal hours. Simultaneous measurement of fasting plasma glucose, insulin, proinsulin, C-peptide, IGF1, and IGF2 during a period of hypoglycemia is the most important diagnostic tool for separating the first clinical type (insulin production) from the second (replacement of liver by tumor) and third (IGF2) types. Proper diagnostic evaluation of cancer patients with fasting hypoglycemia usually will need a 72-hour fast in the hospital with endocrinology consultation.

Management:

- For mild hypoglycemia (glucose level of 50–60 mg/dL), 15 g of simple carbohydrates, such as 4 oz of unsweetened fruit juice or a non-diet soft drink, is sufficient.
- For more severe hypoglycemia without loss of consciousness, 15–20 g of simple carbohydrates should be ingested quickly followed by 15–20 g of a complex carbohydrate, such as crackers or bread.
- For severe hypoglycemia with change in mental status, glucagon (1–2 mg subcutaneously or IV) or glucose (50 mL of 50% dextrose in water IV) should be given promptly.
- The most effective therapeutic approach for non-islet cell tumor-induced hypoglycemia is to resect or debulk the tumor. If unresectable, reducing the tumor bulk via external beam irradiation, intra-arterial chemoembolization, or percutaneous alcohol injection may be attempted.
- Counterregulatory hormones such as glucocorticoids (20–40 mg prednisone equivalents per day) and glucagon (1–2 mg IV or IM) may be administered to raise the blood glucose level. Glucagon infusion (0.5–2 mg/h) to stimulate hepatic gluconeogenesis is an effective therapy for patients with insulin-producing tumors or those with IGF2-mediated hypoglycemia [45].
- A continuous IV infusion of 5–20% dextrose may be required to maintain normal blood glucose in some patients.
- Diazoxide (3–8 mg/kg/d in 2–3 divided doses) has been used successfully to inhibit insulin secretion, but it causes fluid retention, thereby limiting its usefulness at effective doses.
- Treatment of postprandial hypoglycemia is primarily dietary. The diet should have a low carbohydrate content.
 α-Glucosidase inhibitors (acarbose or miglitol) may be helpful.

Tumor Lysis Syndrome

Tumor lysis syndrome (TLS) consists of severe hyperphosphatemia, hyperkalemia, hyperuricemia, azotemia, hypocalcemia, and metabolic acidosis (out of proportion to renal insufficiency) due to the massive release of cell contents and degradation products of dead tumor cells into the bloodstream [46].

Causes:

Factors associated with increased risk of TLS include the following:

 Type of malignancy (e.g., acute lymphocytic leukemia, acute myeloid leukemia with WBC >75,000/µL, Burkitt's lymphoma). TLS can also occur in patients with nonhematologic malignancies, including small cell carcinomas, non-small cell lung cancer, breast cancer, and ovarian cancer.

- Responsiveness to therapy.
- Rapid malignant cell turnover.
- Large tumor burden [47]; pretreatment serum lactate dehydrogenase levels, which tend to correlate with tumor bulk in lymphoma or lymphocytic leukemia may predict the risk of tumor lysis syndrome.
- Preexisting renal insufficiency.
- Acute renal failure shortly after antineoplastic treatment.

Symptoms The symptoms of TLS are nonspecific. Common symptoms include nausea, vomiting, cloudy urine, weakness, fatigue, and arthralgia. Other signs and symptoms related to metabolic and electrolyte abnormalities include neuromuscular irritability, seizures, muscle weakness, and arrhythmia. Arrhythmia may cause sudden death in patients with TLS [48]. Precipitation of uric acid in the renal tubules may lead to nephropathy and acute renal failure [49]. The acute cause of death in TLS is arrhythmia secondary to severe electrolyte abnormalities (especially hyperkalemia) and renal failure. Early recognition of metabolic abnormalities and prompt treatment can avoid fatal outcomes.

Diagnosis TLS can occur spontaneously, but it usually occurs within 72 hours after chemotherapy in patients with leukemia and lymphoma, but new therapeutic regimens may alter the timing of onset. The diagnosis of TLS requires a high level of suspicion because there are few signs or symptoms in the early stage. Routine uric acid and electrolyte screening (including measurement of calcium and phosphorus levels) is indicated in patients with high tumor bulk or hematologic malignancies. The diagnosis of TLS may be based on the Cairo-Bishop definition [47, 50].

Management:

- Once diagnosed, patients with severe TLS should have continuous monitoring of hemodynamic and electrocardiographic parameters in intensive care.
- Management of hyperuricemia:
 - 1. Allopurinol (up to 900 mg/day)
 - Rasburicase (150–200 μg/kg IV daily or one-time dosing with a rescue dose as needed) is a recombinant urate oxidase that converts uric acid to allantoin [51]
 - IV fluid hydration may be coupled with diuresis using loop diuretics (e.g., furosemide, 20–200 mg IV every 4–6 hours) and acetazolamide (250–500 mg IV daily).
 - 4. Urinary alkalinization by sodium bicarbonate or acetate IV infusion to increase the solubility of urate in urine should only be considered in cases of severe hyperuricemia when rasburicase is not available.

- Frequent electrolyte measurements (every 4–6 hours) may be required. See the sections above about management of hyperkalemia, hypocalcemia, and hyperphosphatemia.
- Prompt dialysis should be instituted with continued monitoring until biochemical abnormalities resolve. Indications for dialysis in patients with TLS include the following:
 - 1. Symptomatic hypocalcemia and a serum phosphorus level greater than 3.3 mmol/L (>10.2 mg/dL)
 - 2. Severe azotemia and renal failure (creatinine >10 mg/ dL)
 - 3. Persistent hyperkalemia (>6 mEq/L)
 - 4. Severe hyperuricemia (> 10 mg/dL)
 - 5. Oliguria or anuria despite diuretic use
 - 6. Refractory academia
 - 7. Volume overload

Endocrine Emergencies

Cushing Syndrome

Inappropriate secretion of ACTH, although uncommon, is an important cause of morbidity and mortality in certain types of malignancies. There are at least two different mechanisms: ectopic ACTH production or ectopic production of CRH, the hypothalamic peptide that normally stimulates ACTH synthesis and release.

Causes:

- The most common cause of ectopic ACTH production is the expression of proopiomelanocortin (POMC) by a tumor, producing melanocyte-stimulating hormone and ACTH. The most common tumor associated with ACTH production is small cell lung cancer (SCLC), although pulmonary carcinoid, medullary thyroid carcinoma, islet cell malignancy, pheochromocytoma, and occasional ganglioneuromas can also produce this hormone.
- The second cause of excessive ACTH production is tumor production of CRH [52]. Ectopic production of this peptide causes a clinical syndrome characterized by pituitary corticotroph hyperplasia leading to adrenal cortical hyperplasia and Cushing syndrome. Identification of excessive CRH production requires that the clinician consider this possibility and measure CRH in blood. Neoplasms that can produce CRH include medullary thyroid carcinoma, paragangliomas, prostate cancer, and islet cell neoplasms.

Symptoms Patients with ectopic ACTH syndrome may present with clinical features of Cushing syndrome—easy bruising, centripetal obesity, muscle wasting, hypertension, diabetes, and metabolic alkalosis predominate. Alternatively,

patients with rapidly growing SCLC may present with a clinical syndrome characterized by wasting, muscle atrophy, profound hypokalemic metabolic alkalosis, and hypertension without the other clinical signs of Cushing syndrome.

Diagnosis The hallmark of ectopic ACTH syndrome is the finding of an elevated plasma ACTH concentration. However, in the differential diagnosis of hypercortisolism with an elevated plasma ACTH concentration, the clinician should consider the possibility of an ACTH-producing pituitary tumor [53]. Differentiation between pituitary ACTH production and ectopic tumor production of ACTH or ectopic CRH production should be performed by a consultant endocrinologist, and therefore will not be discussed in detail here. In brief, the diagnostic evaluation starts with confirmation of hypercortisolism and measuring plasma ACTH, followed by dynamic testing, and may involve MRI of pituitary or petrosal venous sinus sampling.

Management:

- Medical management to inhibit cortisol production:
 1. Metyrapone (1–4 g/d orally).
 - 2. Aminoglutethimide (250 mg orally four times per day with upward titration).
 - 3. Ketoconazole (200-400 mg twice a day orally) [54].
 - 4. Etomidate rapidly inhibits cortisol synthesis at subhypnotic doses [55]. It may be titrated from 0.3 to 4 mg/kg/h IV to normalize serum cortisol in some selected patients.
- Surgical removal or treatment of the tumor with chemotherapeutic agents is the primary therapy for an ACTH- or CRH-producing tumor.
- Patients with rapidly progressive small cell lung cancer and ectopic ACTH syndrome have a unique challenge due to the need to initiate chemotherapy quickly. High susceptibility to opportunistic infections after initiation of chemotherapy will often lead to death or serious morbidity [56]. Laparoscopic adrenalectomy, following normalization of electrolyte abnormalities and hypertension, may rapidly treat hypercortisolism to decrease the risk of infectious complication after chemotherapy.
- Replacement glucocorticoid therapy will be needed after adrenalectomy or during pharmacologic inhibition of cortisol production.
- Prophylactic therapy for opportunistic infections caused by *Pneumocystis carinii* or fungi should be considered if chemotherapy is initiated shortly after normalization of the serum cortisol.

Adrenal Crisis

Cancer patients are at increased risk for adrenal insufficiency.

Causes:

Central Adrenal Insufficiency:

- Radiotherapy is a common cause of insidious development of hypothalamic dysfunction; hormonal deficiency can manifest years after radiation. In general, the rapidity of onset and severity of dysfunction depend on the total dose of radiation and the rate of delivery. The somatotropic axis is the most susceptible while the thyrotropic is the least susceptible [57–60].
- High-dose glucocorticoids may suppress the hypothalamicpituitary-corticotropic axis. In cancer patients who have recently discontinued glucocorticoid therapy, acute stress (usually from infection/sepsis) may precipitate an adrenal crisis.
- Acute central adrenal insufficiency may occur in cancer patients in the following settings:
 - 1. Pituitary apoplexy
 - 2. Autoimmune hypophysitis after starting cancer immunotherapy (especially with immune checkpoint inhibitors, i.e., anti-CTLA4, anti-PD1, or anti-PD-L1 antibodies)
- Metastasis to the hypothalamic region or the pituitary gland is uncommon [61].
- Benign tumors such as pituitary tumors and craniopharyngiomas frequently affect this anatomic region and cause endocrine dysfunction.

Primary Adrenal Insufficiency:

- About 20–30% of patients with bilateral adrenal metastasis will have adrenal insufficiency [62], which occurs when more than 80% of adrenal tissue is destroyed or replaced by metastatic cancer [63].
- Bilateral infectious adrenalitis: Many cancer patients may be immunocompromised. In immunocompromised patients with hematological malignancies or stem cell transplantation, infection of the adrenal glands by cytomegalovirus, mycobacteria, or fungi may lead to adrenal insufficiency.
- Bilateral adrenal hemorrhage (e.g., in coagulopathy and thrombocytopenia).
- Bilateral adrenalectomy (e.g., radical nephrectomy and contralateral adrenalectomy for renal cell carcinoma and bilateral adrenal metastasis).
- Autoimmune adrenalitis (e.g., immune checkpoint inhibitors).
- Drugs that are known to inhibit glucocorticoid synthesis: etomidate [62], ketoconazole, aminoglutethimide, metyrapone, megestrol, and mitotane. At high doses, fluconazole and itraconazole may also inhibit the cytochrome P450-dependent enzymes in glucocorticoid synthesis.

Symptoms The symptoms of adrenal insufficiency include weakness, fatigue, nausea, vomiting, and weight loss. In

patients with chronic primary adrenal failure, hyperpigmentation may occur. Acute adrenal crisis involves hypoglycemia and hypotension. The cachexia and weakness seen in adrenal insufficiency can mimic cancer cachexia observed among end-stage cancer patients. Electrolyte abnormalities due to adrenal insufficiency are difficult to distinguish from poor intake, malnutrition, side effects of chemotherapeutic agents, or paraneoplastic syndromes. Both pituitary apoplexy and hypophysitis may be associated with headache.

Diagnosis The medication history should be reviewed for recent glucocorticoid exposure and medications that may inhibit steroid synthesis. Screening tests include basal 8:00 a.m. plasma cortisol measurement, dynamic testing with 1 μ g of cosyntropin (synthetic ACTH₁₋₂₄) or metyrapone (30 mg/kg given orally overnight), and insulin tolerance testing (insulin-induced hypoglycemia).

Without other evidence of metastatic disease elsewhere, whether an adrenal mass is actually a metastatic tumor is critical information in determining the appropriate antineoplastic therapy. In addition to hormonal evaluation, functional scintigraphy using ¹³¹I-6-iodomethyl-19-*nor*cholesterol (NP-59), CT, and MRI may aid in the diagnosis of a unilateral adrenal mass greater than 2 cm [64, 65]. In immunocompromised patients, the possibility of infection of both adrenal glands with cytomegalovirus, mycobacteria, or fungi should be investigated. A high degree of suspicion for hypopituitarism is recommended for patients given ipilimumab, or perhaps other drugs with similar mechanisms of action.

Management If a cancer patient presents to an E with hemodynamic instability, physicians may have insufficient time to wait for the results of serum cortisol measurement or other tests to evaluate adrenal insufficiency. Under such circumstances, empiric treatment with a stress dose of hydrocortisone should be considered based on risk assessment.

- In the event of circulatory instability, sepsis, emergency surgery, or other major complications, stress dosages of parenteral glucocorticoid should be given (e.g., hydrocortisone succinate 100 mg IV every 8 hours).
- Fludrocortisone (0.05–0.20 mg/day) for mineralocorticoid replacement.
- Treat hypotension with IV normal saline or other crystalloid fluid.
- Treat hypoglycemia immediately if symptomatic. Dextrose 50% in water 50–100 mL IV, followed by D5W IV. If IV access is not quickly available, glucagon (2 mg) may be given subcutaneously or intramuscularly, but the effect may be delayed by about 10–20 minutes.
- If and when the patient is clinically stable, arrangement should be made for endocrinology consultation and the ACTH stimulation test.

Hyperthyroidism

Thyrotoxicosis is a common disease with a prevalence of 20–25 per 100,000 in the general population and a female:male ratio of 5:1.

Causes:

- Graves disease, toxic multinodular goiters, and solitary toxic nodules are the three forms of primary hyperthyroidism that account for most cases of hyperthyroidism in the general population. The risk of Graves disease after radiotherapy for Hodgkin disease is estimated to be at least 7.2 times that in the general population [66]. Immune checkpoint inhibitors may induce Graves disease in rare occasions.
- Large quantities of iodide are present in many drugs (e.g., approximately 9 mg of iodine following a daily amiodarone dose of 300 mg), antiseptics (e.g., povidone-iodine), and contrast media used in radiology. Iodine-induced hyperthyroidism usually occurs in patients with underlying thyroid diseases.
- Autoimmune thyroiditis may be precipitated by bioimmunotherapy for cancer with cytokines or immune checkpoint inhibitors. In addition to being a source of excess iodide described above, amiodarone may induce thyroiditis. Transient hyperthyroidism is usually followed by hypothyroidism.
- Radiation-induced painless thyroiditis with hyperthyroxinemia is an uncommon side effect of external beam radiotherapy to the head and neck area. Transient hyperthyroidism is also followed by hypothyroidism.
- Thyroid metastasis occurs in 1.25–24.00% of patients with metastatic carcinoma; however, thyrotoxicosis owing to follicular destruction by metastasis is rare.
- Structural homology in the human chorionic gonadotropin and TSH molecules as well as receptors provides the biochemical basis for the ability of human chorionic gonadotropin to stimulate the TSH receptor. Trophoblastic tumors, hydatidiform moles, and choriocarcinomas secrete human chorionic gonadotropin in large amounts, often causing hyperthyroidism. Hyperthyroidism is likely for human chorionic gonadotropin level >200 IU/mL.

Removal or effective therapy for the underlying tumor is the most effective therapy for clinical syndromes caused by excessive β -HCG production. Hyperthyroidism can be treated short term with thionamide therapy if there is belief that chemotherapy or other strategies to treat the underlying malignancy are likely to be effective. In patients with less responsive tumors, thyroidectomy or radioactive iodine may be required.

Symptoms Thyrotoxicosis is characterized by a hyperadrenergic state. Sinus tachycardia, systolic flow murmur, and water-hammer pulse are common. Atrial dysrhythmias (atrial
fibrillation, atrial flutter, and premature atrial contractions) and congestive heart failure are often observed. Eye signs include exophthalmos, lid lag, and upper lid retraction. Neuropsychiatric symptoms include agitation, anxiety, restlessness, fear, paranoia, and mood swings. Neuromuscular symptoms include fine tremor in the hands and proximal myopathy (common in the elderly). Gastrointestinal symptoms include hyperphagia, diarrhea, nausea, vomiting, and abdominal pain. Skin signs include flushed skin, hair loss, and pretibial myxedema. Apathetic hyperthyroidism is seen in the elderly with prominent features of congestive heart failure, atrial fibrillation, and weight loss.

Diagnosis Thyrotoxicosis is diagnosed by measuring thyroid hormone (thyroxin and triiodothyronine) and TSH levels. Measurement of free thyroid hormones instead of total serum hormone prevents changes introduced by variations in thyroxine-binding globulin. Pituitary and hypothalamic causes of thyrotoxicosis are very rare. Measurements of thyroid-stimulating immunoglobulin and anti-thyroperoxidase antibodies are helpful in evaluating autoimmune etiologies. A radionuclide scan is helpful in distinguishing hyperfunction of the thyroid gland from thyroiditis.

Thyroid storm should be considered in the differential diagnosis of hyperpyrexia in the emergency care setting, particularly in cancer patients with risk factors for Graves disease (e.g., bioimmunotherapy and history of irradiation of the neck or chest area) or tumors that may secrete human chorionic gonadotropin. A scoring system for thyroid storm and a set of diagnostic criteria, including fever, tachycardia, tachyarrhythmias, and mental status changes, have been proposed [67, 68].

Management:

- Treatment of Graves disease includes anti-thyroid medications, radioactive iodine, and surgery.
- Treatment of thyroiditis primarily involves removing the causative factors and controlling the hyperadrenergic symptoms with β-blockers.
- If thyroid storm is highly likely on the basis of clinical criteria, diagnostic studies should be performed, and therapy should be initiated immediately. In addition to support for systemic decompensation and correction of precipitating factors, acute management may involve the following:
 - Propylthiouracil 100–600 mg/day or methimazole 10–60 mg/day.
 - β-blockers, both cardioselective and noncardioselective, are important adjuncts in treating hyperthyroidism.β-blockade provides rapid relief of hyperadrenergic symptoms and signs of thyrotoxicosis. High doses of propranolol (greater than 160 mg/day) can inhibit peripheral conversion of T4 to T3.

- Saturated solution potassium iodide (3–5 drops) is administered orally every 8 hours to block release of thyroid hormones in patients with thyrotoxicosis. At pharmacologic concentrations (100 times the normal plasma level), iodides decrease thyroid gland activity through the Wolff-Chaikoff effect.
- The oral contrast agents also are potent inhibitors of T4-to-T3 conversion, making them ideal for treatment of severe or decompensated thyrotoxicosis. They are generally given after starting treatment with thioamide. Although physicians have used IV iodinated radiographic contrast medium to treat a case of thyroid storm, this approach is highly nephrotoxic, and its efficacy has yet to be firmly established.
- Other treatment options include corticosteroids (e.g., dexamethasone, which inhibits peripheral thyroxine conversion), colestipol, lithium, amiodarone, ipodate, iopanoic acid, and potassium perchlorate.
- Plasmapheresis and hemoperfusion are effective ways to remove excess thyroid hormone.

Myxedema Coma

The prevalence of hypothyroidism is 2-3% in the general population with a female-to-male ratio of 10:1. Therefore, female cancer patients with preexisting or co-existing hypothyroidism are common. Hypothyroidism may also be a complication of cancer or its treatment.

Causes:

- Total or near-total thyroidectomy may be performed for a variety of oncologic reasons in the management of thyroid cancer, head and neck cancer, or thyroid metastasis. Thyroid replacement is needed in this group of patients.
- Irradiation can cause primary, secondary, or tertiary hypothyroidism.
- Primary hypothyroidism is caused by thyroid cell destruction, inhibition of cell division, vascular damage, and possibly an immune-mediated phenomenon. Factors that increase the risk of developing primary hypothyroidism include a high radiation dose to the vicinity of the thyroid gland, duration since therapy, lack of shielding of the thyroid during therapy, and combined irradiation and surgical treatments [69].
- Hypothyroidism after radiation therapy is related to the radiation dose. The threshold for causing clinical hypothyroidism is about 10 Gy [66, 70].
- Chemotherapy
 - The incidence of primary hypothyroidism is increased in patients treated with multiple combination drug regimens [71, 72], with or without radiation [71].

- L-Asparaginase, in addition to inhibition of TBG synthesis discussed above, may also inhibit TSH synthesis reversibly and lead to temporary hypothyroidism with decreased free T₄ levels [73].
- Immunotherapy
 - Thyroid dysfunction is a recognized side effect of cytokine treatments. Treatment with interleukin-2 produces thyroid dysfunction in approximately 20–35% of patients [74]. These patients have hypothyroidism, hyperthyroidism, or hyperthyroidism followed by hypothyroidism [75]. Approximately 10% of interferon-treated patients develop primary hypothyroidism [76]. Patients with anti-thyroid antibodies before therapy are at higher risk of cytokine-induced thyroid dysfunction.
 - Immune checkpoint inhibitors induce immune thyroiditis. In the late phase of thyroiditis, long-term or permanent hypothyroidism results.
- Targeted therapy
 - Bexarotene (a retinoid X receptor [RXR]-selective ligand to treat cutaneous T-cell lymphoma) caused secondary hypothyroidism dose-dependently [77]. In addition to suppressing transcription of TSH by an RXR-mediated thyroid hormone-independent mechanism [78], bexarotene also increases clearance of thyroid hormones by a metabolic pathway not involving deiodinase [79].
 - Many TKIs (e.g., pazopanib, nilotinib, axitinib, cabozantinib, sorafenib, dasatinib, sunitinib, and imatinib) may cause primary hypothyroidism. Tyrosine kinase receptors (e.g., VEGFR, EGFR, RET, KIT, MET) and downstream signaling pathways (e.g., BRAF, PI3K, and mTOR pathways), especially RET and BRAF, play important roles in thyroid physiology. Imatinib and sunitinib may have additional mechanisms for inhibition of thyroid function. These inhibitors inhibit different targets with varying potencies/selectivity, and relative contribution of inhibition of specific pathways or specific etiologic mechanisms may vary.
 - Using high-dose (100–1000 mCi) [¹³¹I]-metaiodobenzylguanidine to treat unresectable pheochromocytoma may result in primary hypothyroidism.

Symptoms Hypothyroid symptoms are nonspecific and include fatigue, general weakness, cold intolerance, depression, weight gain, joint aches, constipation, dry skin, and menstrual irregularities. Signs of moderate to severe hypothyroidism include hypertension, bradycardia, coarse hair, periorbital edema, carpal tunnel syndrome, and delayed relaxation of the tendon reflexes. Unusual signs of severe

hypothyroidism include megacolon, cardiomegaly, and congestive heart failure.

Myxedema coma may occur in patients with hypothyroidism and be life-threatening as the severity of hypothermia, bradycardia, and hypoventilation increases. Pericardial, pleural, and peritoneal effusions are often present. An ileus is present in about two thirds of cases. Central nervous system changes in these patients include seizures, stupor, and coma.

Diagnosis The diagnosis of hypothyroidism is confirmed by thyroid function tests. Myxedema coma is rare since screening for thyroid dysfunction in cancer patients on targeted therapy or immunotherapy is routine. In most cases, TSH and free T4 testing is adequate for initial evaluation. In patients with myxedema coma, serum thyroid hormone levels are usually very low, whereas the TSH level is quite high (except in cases of secondary hypothyroidism).

Anemia, hyponatremia, hypoglycemia, hypothermia, and hypotension can occur. Arterial blood gas measurement usually reveals retention of carbon dioxide and hypoxemia. An EKG often shows sinus bradycardia, various types and degrees of heart block, low voltage, and T-wave flattening.

Management Recognition of hypothyroidism may be difficult in the emergency care setting. Thyroid function test results typically are not available expeditiously. The emergency physician's responsibility is to consider the diagnosis of hypothyroidism, provide acute care, and order the appropriate thyroid function tests to expedite diagnosis. Myxedema coma occurs most often in elderly hypothyroidism patients with a superimposed precipitating event, which must also be treated.

Rapid clinical diagnosis with early therapy may be lifesaving. In critically ill patients, if free thyroxine is very low and myxedema coma is highly suspected, 0.3–0.5 mg of levothyroxine should be given IV once, followed by 0.025– 0.100 mg/day. Other supportive measures, such as correction of hypothermia using slow rewarming and ventilatory and circulatory support, are critical.

References

- 1. Flombaum CD. Metabolic emergencies in the cancer patient. Semin Oncol. 2000;27(3):322–34.
- 2. Johnson BE, Damodaran A, Rushin J, Gross A, Le PT, Chen HC, et al. Ectopic production and processing of atrial natriuretic peptide in a small cell lung carcinoma cell line and tumor from a patient with hyponatremia. Cancer. 1997;79(1):35–44.
- Bode U, Seif SM, Levine AS. Studies on the antidiuretic effect of cyclophosphamide: vasopressin release and sodium excretion. Med Pediatr Oncol. 1980;8(3):295–303.
- Anand AJ, Bashey B. Newer insights into cisplatin nephrotoxicity. Ann Pharmacother. 1993;27(12):1519–25.

- 5. Raftopoulos H. Diagnosis and management of hyponatremia in cancer patients. Support Care Cancer. 2007;15(12):1341–7.
- Seelig MS. Magnesium deficiency and cardiac dysrhythmia. In: Magnesium deficiency in the pathogenesis of disease. Topics in bone and mineral disorders. Boston MA: Springer; 1980. p. 222. https://doi.org/10.1007/978-1-4684-9108-1_9.
- Arbaiza D, Noriega K, Marcial J, Wachtel A, Perez C, Torres CF. Ectopic production of prolactin in an infant with non-Hodgkin lymphoma. Med Pediatr Oncol. 1999;32(4):311–2.
- Stewart AF, Keating T, Schwartz PE. Magnesium homeostasis following chemotherapy with cisplatin: a prospective study. Am J Obstet Gynecol. 1985;153(6):660–5.
- Haroon N, Raza SM, Bhat ZY. Hypomagnesemia and chemotherapy, diagnostic dilemma, and treatment challenge: case report and literature review. Am J Ther. 2016;23(4):e1085–90.
- Wang Q, Qi Y, Zhang D, Gong C, Yao A, Xiao Y, et al. Electrolyte disorders assessment in solid tumor patients treated with anti-EGFR monoclonal antibodies: a pooled analysis of 25 randomized clinical trials. Tumour Biol. 2015;36(5):3471–82.
- Vassilopoulou-Sellin R, Newman BM, Taylor SH, Guinee VF. Incidence of hypercalcemia in patients with malignancy referred to a comprehensive cancer center. Cancer. 1993;71(4):1309–12.
- Stewart AF. Clinical practice. Hypercalcemia associated with cancer. N Engl J Med. 2005;352(4):373–9.
- Roodman GD. Mechanisms of bone metastasis. N Engl J Med. 2004;350(16):1655–64.
- Seymour JF, Gagel RF, Hagemeister FB, Dimopoulos MA, Cabanillas F. Calcitriol production in hypercalcemic and normocalcemic patients with non-Hodgkin lymphoma. Ann Intern Med. 1994;121(9):633–40.
- Cohen J, Gierlowski TC, Schneider AB. A prospective study of hyperparathyroidism in individuals exposed to radiation in childhood. JAMA. 1990;264(5):581–4.
- Body JJ, Bartl R, Burckhardt P, Delmas PD, Diel IJ, Fleisch H, et al. Current use of bisphosphonates in oncology. International Bone and Cancer Study Group. J Clin Oncol. 1998;16(12):3890–9.
- Berenson JR, Rosen LS, Howell A, Porter L, Coleman RE, Morley W, et al. Zoledronic acid reduces skeletal-related events in patients with osteolytic metastases. Cancer. 2001;91(7):1191–200.
- Bekker PJ, Holloway DL, Rasmussen AS, Murphy R, Martin SW, Leese PT, et al. A single-dose placebo-controlled study of AMG 162, a fully human monoclonal antibody to RANKL, in postmenopausal women. J Bone Miner Res. 2004;19(7):1059–66.
- Norton JA, Venzon DJ, Berna MJ, Alexander HR, Fraker DL, Libutti SK, et al. Prospective study of surgery for primary hyperparathyroidism (HPT) in multiple endocrine neoplasia-type 1 and Zollinger-Ellison syndrome: long-term outcome of a more virulent form of HPT. Ann Surg. 2008;247(3):501–10.
- Yeung SC, Chiu AC, Vassilopoulou-Sellin R, Gagel RF. The endocrine effects of nonhormonal antineoplastic therapy. Endocr Rev. 1998;19(2):144–72.
- Jonsson KB, Zahradnik R, Larsson T, White KE, Sugimoto T, Imanishi Y, et al. Fibroblast growth factor 23 in oncogenic osteomalacia and X-linked hypophosphatemia. N Engl J Med. 2003;348(17):1656–63.
- Nuovo MA, Dorfman HD, Sun CC, Chalew SA. Tumor-induced osteomalacia and rickets. Am J Surg Pathol. 1989;13(7):588–99.
- Brade WP, Herdrich K, Kachel-Fischer U, Araujo CE. Dosing and side-effects of ifosfamide plus mesna. J Cancer Res Clin Oncol. 1991;117(Suppl 4):S164–86.
- 24. Moncrieff M, Foot A. Fanconi syndrome after ifosfamide. Cancer Chemother Pharmacol. 1989;23(2):121–2.
- Citrin DL, Wallemark CB, Nadler R, Geiger C, Tuttle K, Kaplan EH, et al. Estramustine affects bone mineral metabolism in metastatic prostate cancer. Cancer. 1986;58(10):2208–13.

- Kumar R. Tumor-induced osteomalacia and the regulation of phosphate homeostasis. Bone. 2000;27(3):333–8.
- Nilsen TI, Vatten LJ. Prospective study of colorectal cancer risk and physical activity, diabetes, blood glucose and BMI: exploring the hyperinsulinaemia hypothesis. Br J Cancer. 2001;84(3):417–22.
- Muti P, Quattrin T, Grant BJ, Krogh V, Micheli A, Schunemann HJ, et al. Fasting glucose is a risk factor for breast cancer: a prospective study. Cancer Epidemiol Biomark Prev. 2002;11(11):1361–8.
- Verlato G, Zoppini G, Bonora E, Muggeo M. Mortality from sitespecific malignancies in type 2 diabetic patients from Verona. Diabetes Care. 2003;26(4):1047–51.
- Richardson LC, Pollack LA. Therapy insight: influence of type 2 diabetes on the development, treatment and outcomes of cancer. Nat Clin Pract Oncol. 2005;2(1):48–53.
- Coughlin SS, Calle EE, Teras LR, Petrelli J, Thun MJ. Diabetes mellitus as a predictor of cancer mortality in a large cohort of US adults. Am J Epidemiol. 2004;159(12):1160–7.
- 32. Yamagata H, Kiyohara Y, Nakamura S, Kubo M, Tanizaki Y, Matsumoto T, et al. Impact of fasting plasma glucose levels on gastric cancer incidence in a general Japanese population: the Hisayama study. Diabetes Care. 2005;28(4):789–94.
- 33. van de Poll-Franse LV, Houterman S, Janssen-Heijnen ML, Dercksen MW, Coebergh JW, Haak HR. Less aggressive treatment and worse overall survival in cancer patients with diabetes: a large population based analysis. Int J Cancer. 2007;120(9):1986–92.
- Schein PS, O'Connell MJ, Blom J, Hubbard S, Magrath IT, Bergevin P, et al. Clinical antitumor activity and toxicity of streptozotocin (NSC-85998). Cancer. 1974;34(4):993–1000.
- Gillette PC, Hill LL, Starling KA, Fernbach DJ. Transient diabetes mellitus secondary to L-asparaginase therapy in acute leukemia. J Pediatr. 1972;81(1):109–11.
- Almawi WY, Tamim H, Azar ST. Clinical review 103: T helper type 1 and 2 cytokines mediate the onset and progression of type I (insulin-dependent) diabetes. J Clin Endocrinol Metab. 1999;84(5):1497–502.
- Nunnery SE, Mayer IA. Management of toxicity to isoform alphaspecific PI3K inhibitors. Ann Oncol. 2019;30(Suppl 10):x21–x6.
- Verges B, Walter T, Cariou B. Endocrine side effects of anti-cancer drugs: effects of anti-cancer targeted therapies on lipid and glucose metabolism. Eur J Endocrinol. 2014;170(2):R43–55.
- 39. Drachenberg CB, Klassen DK, Weir MR, Wiland A, Fink JC, Bartlett ST, et al. Islet cell damage associated with tacrolimus and cyclosporine: morphological features in pancreas allograft biopsies and clinical correlation. Transplantation. 1999;68(3):396–402.
- Jindal RM, Sidner RA, Milgrom ML. Post-transplant diabetes mellitus. The role of immunosuppression. Drug Saf. 1997;16(4):242–57.
- 41. Baxter RC, Holman SR, Corbould A, Stranks S, Ho PJ, Braund W. Regulation of the insulin-like growth factors and their binding proteins by glucocorticoid and growth hormone in nonislet cell tumor hypoglycemia. J Clin Endocrinol Metab. 1995;80(9):2700–8.
- Baxter RC. Insulin-like growth factor binding proteins as glucoregulators. Metabolism. 1995;44(10 Suppl 4):12–7.
- Baxter RC. The role of insulin-like growth factors and their binding proteins in tumor hypoglycemia. Horm Res. 1996;46(4–5):195–201.
- 44. Sillos EM, Shenep JL, Burghen GA, Pui CH, Behm FG, Sandlund JT. Lactic acidosis: a metabolic complication of hematologic malignancies: case report and review of the literature. Cancer. 2001;92(9):2237–46.
- Hoff AO, Vassilopoulou-Sellin R. The role of glucagon administration in the diagnosis and treatment of patients with tumor hypoglycemia. Cancer. 1998;82(8):1585–92.
- Howard SC, Jones DP, Pui CH. The tumor lysis syndrome. N Engl J Med. 2011;364(19):1844–54.
- 47. Wilson FP, Berns JS. Tumor lysis syndrome: new challenges and recent advances. Adv Chronic Kidney Dis. 2014;21(1):18–26.

- Van Der Klooster JM, Van Der Wiel HE, Van Saase JL, Grootendorst AF. Asystole during combination chemotherapy for non-Hodgkin's lymphoma: the acute tumor lysis syndrome. Neth J Med. 2000;56(4):147–52.
- 49. Hsu HH, Chen YC, Tian YC, Chan YL, Kuo MC, Tang CC, et al. Role of serum sodium in assessing hospital mortality in cancer patients with spontaneous tumour lysis syndrome inducing acute uric acid nephropathy. Int J Clin Pract. 2009;63(5):751–6.
- 50. Cairo MS, Bishop M. Tumour lysis syndrome: new therapeutic strategies and classification. Br J Haematol. 2004;127(1):3–11.
- Mahmoud HH, Leverger G, Patte C, Harvey E, Lascombes F. Advances in the management of malignancy-associated hyperuricaemia. Br J Cancer. 1998;77(Suppl 4):18–20.
- 52. O'Brien T, Young WF Jr, Davila DG, Scheithauer BW, Kovacs K, Horvath E, et al. Cushing's syndrome associated with ectopic production of corticotrophin-releasing hormone, corticotrophin and vasopressin by a phaeochromocytoma. Clin Endocrinol. 1992;37(5):460–7.
- Newell-Price J, Trainer P, Besser M, Grossman A. The diagnosis and differential diagnosis of Cushing's syndrome and pseudo-Cushing's states. Endocr Rev. 1998;19(5):647–72.
- Winquist EW, Laskey J, Crump M, Khamsi F, Shepherd FA. Ketoconazole in the management of paraneoplastic Cushing's syndrome secondary to ectopic adrenocorticotropin production. J Clin Oncol. 1995;13(1):157–64.
- 55. Schulte HM, Benker G, Reinwein D, Sippell WG, Allolio B. Infusion of low dose etomidate: correction of hypercortisolemia in patients with Cushing's syndrome and dose-response relationship in normal subjects. J Clin Endocrinol Metab. 1990;70(5):1426–30.
- 56. Dimopoulos MA, Fernandez JF, Samaan NA, Holoye PY, Vassilopoulou-Sellin R. Paraneoplastic Cushing's syndrome as an adverse prognostic factor in patients who die early with small cell lung cancer. Cancer. 1992;69(1):66–71.
- 57. Lam KS, Tse VK, Wang C, Yeung RT, Ho JH. Effects of cranial irradiation on hypothalamic-pituitary function--a 5-year longitudinal study in patients with nasopharyngeal carcinoma. Q J Med. 1991;78(286):165–76.
- 58. Pai HH, Thornton A, Katznelson L, Finkelstein DM, Adams JA, Fullerton BC, et al. Hypothalamic/pituitary function following high-dose conformal radiotherapy to the base of skull: demonstration of a dose-effect relationship using dose-volume histogram analysis. Int J Radiat Oncol Biol Phys. 2001;49(4):1079–92.
- Samaan NA, Schultz PN, Yang KP, Vassilopoulou-Sellin R, Maor MH, Cangir A, et al. Endocrine complications after radiotherapy for tumors of the head and neck. J Lab Clin Med. 1987;109(3):364–72.
- Shalet SM. Disorders of the endocrine system due to radiation and cytotoxic chemotherapy. Clin Endocrinol. 1983;19(5):637–59.
- Fassett DR, Couldwell WT. Metastases to the pituitary gland. Neurosurg Focus. 2004;16(4):E8.
- Redman BG, Pazdur R, Zingas AP, Loredo R. Prospective evaluation of adrenal insufficiency in patients with adrenal metastasis. Cancer. 1987;60(1):103–7.

- Cedermark BJ, Sjoberg HE. The clinical significance of metastases to the adrenal glands. Surg Gynecol Obstet. 1981;152(5):607–10.
- Francis IR, Smid A, Gross MD, Shapiro B, Naylor B, Glazer GM. Adrenal masses in oncologic patients: functional and morphologic evaluation. Radiology. 1988;166(2):353–6.
- Hussain S, Belldegrun A, Seltzer SE, Richie JP, Abrams HL. CT diagnosis of adrenal abnormalities in patients with primary nonadrenal malignancies. Eur J Radiol. 1986;6(2):127–31.
- Hancock SL, Cox RS, McDougall IR. Thyroid diseases after treatment of Hodgkin's disease. N Engl J Med. 1991;325(9):599–605.
- Burch HB, Wartofsky L. Life-threatening thyrotoxicosis. Thyroid storm. Endocrinol Metab Clin N Am. 1993;22(2):263–77.
- Wartofsky L. Clinical criteria for the diagnosis of thyroid storm. Thyroid. 2012;22(7):659–60.
- Tami TA, Gomez P, Parker GS, Gupta MB, Frassica DA. Thyroid dysfunction after radiation therapy in head and neck cancer patients. Am J Otolaryngol. 1992;13(6):357–62.
- Schimpff SC, Diggs CH, Wiswell JG, Salvatore PC, Wiernik PH. Radiation-related thyroid dysfunction: implications for the treatment of Hodgkin's disease. Ann Intern Med. 1980;92(1):91–8.
- Sutcliffe SB, Chapman R, Wrigley PF. Cyclical combination chemotherapy and thyroid function in patients with advanced Hodgkin's disease. Med Pediatr Oncol. 1981;9(5):439–48.
- Ogilvy-Stuart AL, Shalet SM, Gattamaneni HR. Thyroid function after treatment of brain tumors in children. J Pediatr. 1991;119(5):733–7.
- Heidemann PH, Stubbe P, Beck W. Transient secondary hypothyroidism and thyroxine binding globulin deficiency in leukemic children during polychemotherapy: an effect of L-asparaginase. Eur J Pediatr. 1981;136(3):291–5.
- 74. Krouse RS, Royal RE, Heywood G, Weintraub BD, White DE, Steinberg SM, et al. Thyroid dysfunction in 281 patients with metastatic melanoma or renal carcinoma treated with interleukin-2 alone. J Immunother Emphasis Tumor Immunol. 1995;18(4):272–8.
- Vassilopoulou-Sellin R, Sella A, Dexeus FH, Theriault RL, Pololoff DA. Acute thyroid dysfunction (thyroiditis) after therapy with interleukin-2. Horm Metab Res. 1992;24(9):434–8.
- 76. Baudin E, Marcellin P, Pouteau M, Colas-Linhart N, Le Floch JP, Lemmonier C, et al. Reversibility of thyroid dysfunction induced by recombinant alpha interferon in chronic hepatitis C. Clin Endocrinol. 1993;39(6):657–61.
- Sherman SI. Etiology, diagnosis, and treatment recommendations for central hypothyroidism associated with bexarotene therapy for cutaneous T-cell lymphoma. Clin Lymphoma. 2003;3(4):249–52.
- Golden WM, Weber KB, Hernandez TL, Sherman SI, Woodmansee WW, Haugen BR. Single-dose rexinoid rapidly and specifically suppresses serum thyrotropin in normal subjects. J Clin Endocrinol Metab. 2007;92(1):124–30.
- Smit JW, Stokkel MP, Pereira AM, Romijn JA, Visser TJ. Bexarotene-induced hypothyroidism: bexarotene stimulates the peripheral metabolism of thyroid hormones. J Clin Endocrinol Metab. 2007;92(7):2496–9.



Amit Lahoti

Case Study

A 70-year-old woman with a history of ovarian cancer s/p debulking surgery was started on chemotherapy with carboplatin and paclitaxel. Bevacizumab was added to the last 6 chemotherapy cycles. The patient developed relative thrombocytopenia with a drop in the platelet count from 276 to 71 K/µL. Nifedipine XL 60 mg PO daily was added to bisoprolol 10 mg PO daily due to worsening blood pressure. The patient was subsequently evaluated in the emergency department due to elevated blood pressure (BP 216/93) and worsening renal function. Serum creatinine increased from a baseline of 1.2–1.9 mg/dL. She was treated with clonidine and hydralazine and admitted to the hospital with a presumptive diagnosis of drug-induced thrombotic microangiopathy due to bevacizumab (anti-VEGF therapy) and further treatment was withheld. However, the serum creatinine continued to increase over the next 5 months to 4.5 mg/dL. A renal biopsy was performed at that time which confirmed acute thrombotic microangiopathy (TMA). Since she had persistent disease despite stopping bevacizumab therapy, she was started on eculizumab, a terminal complement inhibitor. After 6 months of therapy, the serum creatinine stabilized at 2.6 mg/dL. Platelet count recovered to 273 K/uL. She has remained off of eculizumab for over 1 year with no evidence of recurrence of TMA.

Chapter Overview

Renal emergencies are common during the care of patients with cancer and require a multidisciplinary approach between the emergency physician, nephrologist, oncologist, and interventional radiologist. Acute kidney injury (AKI) is a frequent complication of cancer treatment associated with a higher mortality rate and hospital length of stay. General causes of AKI such as acute tubular necrosis and hypovolemia are still common in patients with cancer. However, there are other etiologies of AKI that are more specific to cancer, such as AKI in the setting of multiple myeloma, stem cell transplant, tumor lysis syndrome, targeted therapy, and immunotherapy. Patients with cancer may also present with severe electrolyte derangements requiring immediate treatment by the emergency physician.

Introduction

The kidneys are important in regulating electrolyte and acidbase levels, eliminating waste products and fluid, and producing enzymes and hormones. Compared to other organs, the kidneys receive the highest amount of blood supply from the heart on a per-gram basis. Therefore, the kidneys are vulnerable to AKI from toxins, drugs, or metabolites circulating in the bloodstream. Advances in conventional chemotherapy, targeted therapy, and immunotherapy have improved the overall survival of patients with cancer. However, many of these therapies entail a higher risk of AKI. Prevention is essential since there is no effective therapy available for AKI. This chapter emphasizes some of the more common renal problems that emergency physicians encounter during cancer treatment.

Acute Kidney Injury in Cancer Patients

The incidence of AKI in the setting of cancer varies with the underlying population, tumor type, and treatment. A 7-year study in Denmark examining the incidence of AKI among 37,267 patients with cancer reported a rate of 17.5% within 1 year of cancer diagnosis [1]. Long-term dialysis was required in 5.1% of patients developing AKI. However, other studies involving higher risk, critically ill patients report the

A. Lahoti (🖂)

Department of Nephrology, The University of Texas MD Anderson Cancer Center, Houston, TX, USA e-mail: alahoti@mdanderson.org

need for dialysis in up to 60% of patients with AKI. Patients with renal cell carcinoma, hepatocellular carcinoma, multiple myeloma, lymphoma, and leukemia were at highest risk of AKI. A Canadian study between 2007 and 2014 found the overall cumulative incidence of AKI among 163,071 patients undergoing cancer treatment to be 9.3% [2]. Patients with myeloma, bladder cancer, and leukemia were at highest risk. The annual incidence of AKI also increased from 18 to 52 per 1000 person-years during that time period. Development of AKI correlates with increased hospital length of stay, health-care costs, and mortality rates [3]. AKI increases the toxic effects of chemotherapy, excludes patients from clinical trials, and limits further cancer treatment. The causes of AKI are generally classified into three main categories: renal hypoperfusion, intrinsic renal disease, and postrenal obstruction (Table 30.1).

Varying definitions of AKI have been historically used in patient care and clinical research. In 2004, a standard classification termed the Risk, Injury, Failure, Loss, End-Stage (RIFLE) criteria was proposed based on relative rise in serum creatinine (SCr) or progressive oliguria [4]. A 50% rise in SCr from baseline defined the earliest stage of AKI. The criteria were recently modified by the Kidney Disease Improving Global Outcomes (KDIGO) AKI Workgroup to include even smaller increases in SCr (>= 0.3 mg/dL) as well as a time constraint for the rise in creatinine elevation (Table 30.2). While decreased urine output is also part of the definition, it is often cumbersome to measure and not used as routinely. The KDIGO criteria for AKI are now universally accepted both in clinical practice and research. AKI as defined by RIFLE or KDIGO has been validated as a negative prognostic marker [5, 6]. In a study of patients with newly diagnosed AML undergoing induction chemotherapy, there was a stepwise increase in mortality associated with the no-AKI, Risk, Injury, and Failure categories (3.8%, 13.6%, 19.6%, and 61.7%, respectively) [7]. Another study of critically ill patients with cancer found a 1.3-, 3.0-, and 14-fold increase in 60-day mortality for the Risk, Injury, and Failure categories, respectively [3]. More recent studies using the KDIGO definition have found a similar stepwise increase in mortality with progressive AKI [8].

A brief clinical exam, routine labs, and optimization of hemodynamics are necessary for patients presenting with AKI upon arrival to the emergency department (ED). Volume depletion can manifest by orthostatic hypotension, tachycardia, poor skin turgor, dry mucous membranes, and low central venous pressure. Intravenous (IV) hydration, preferably with crystalloid solutions, should be given to target a mean arterial pressure greater than 65 mmHg. Patients with prerenal azotemia may have a blood urea nitrogen-SCr ratio greater than 20, a fractional excretion of sodium less than 1%, a urine sodium level less than 20 mEq/L, and the presence of hyaline casts. A fractional excretion of sodium

 Table 30.1
 Common causes of acute kidney injury in patients with cancer

ancer
Renal hypoperfusion
Volume depletion
Nausea, vomiting, diarrhea
Decreased oral intake due to mucositis (5-fluorouracil, methotrexate, taxanes)
Polyuria caused by hyperglycemia (steroids) or diabetes insipidus (pituitary tumor)
"Third spacing" (hypoalbuminemia, liver or peritoneal metastases, interleukin 2)
Insensible loss of fluid from skin lesions (mycosis fungoides)
Hemodynamic mediated
Sepsis
Renal arteriolar vasoconstriction (NSAIDs, calcineurin inhibitors, hypercalcemia)
Congestive heart failure
Hepatorenal syndrome/hepatic sinusoidal obstruction syndrome
Budd-Chiari syndrome
caused by hepatomegaly or a tumor
IV iodinated contrast agent
Abdominal compartment syndrome
Intrinsic renal disease
Acute tubular necrosis
Chemotherapy (cisplatin, ifosfamide)
Anti-infectives (amphotericin B, foscarnet, cidofovir, aminoglycosides, vancomycin)
Bisphosphonates
Sepsis
Prolonged prerenal azotemia
Acute interstitial nephritis (penicillins, cephalosporins, fluoroquinolones, NSAIDs, checkpoint inhibitors)
Crystal nephropathy (methotrexate, acyclovir, ciprofloxacin, sulfonamides, rifampin)
Osmotic nephrosis (IV immunoglobulin, mannitol, starch)
Thrombotic microangiopathy (post-HSCT, gemcitabine, anti- VEGF therapy, prior radiation therapy)
Myeloma-related kidney disease
Tumor lysis syndrome
Tumor infiltration of the kidney
Glomerulonephritis
Lysozymuria (CMML or AML) with direct tubular injury
Cytokine release syndrome (engraftment syndrome, CAR T-cell therapy)
Postrenal obstruction
Bladder outlet obstruction (malignancy of the cervix, prostate, bladder, or uterus)
Retroperitoneal disease (metastasis, lymphadenopathy, fibrosis)
Hemorrhagic cystitis (cyclophosphamide, BK* virus, adenovirus)
Ureteral strictures (prior radiation therapy, BK virus)
ISCT hematopoietic stem cell transplantation VEGE vascular endo
is criminatopolene stem een nansplantation, rEGT vaschial chuo

HSCT hematopoietic stem cell transplantation, *VEGF* vascular endothelial growth factor, *CMML* chronic myelomonocytic leukemia, *AML* acute myeloid leukemia, *CAR* chimeric antigen receptor ^aInitials of the patient in whom it was first detected in 1971

greater than 2%, a urine sodium level greater than 40 mEq/L, and the presence of coarse granular casts are suggestive of acute tubular necrosis. Acute interstitial nephritis is a common cause of AKI in patients with cancer given the high

 Table 30.2
 Kidney Disease Improving Global Outcomes (KDIGO)

 criteria for AKI (https://kdigo.org/guidelines/acute-kidney-injury/)

KDIGO stage (corresponding RIFLE stage) 1 (Risk)	Increase in creatinine level ≥50% from baseline within 7 days or ≥0.3 mg/dL within 48 h	Decrease in urine output <0.5 mL/kg/h × 6 h
2 (Injury)	$\geq 100\%$ from baseline	<0.5 mL/kg/h × 12 h
3 (Failure)	\geq 200% from baseline, absolute SCr level \geq 4 mg/dL with acute rise \geq 0.5 mg/dL, or need for dialysis	<0.3 mL/kg/h × 24 h or anuria × 12 h

usage of antibiotics, NSAIDs, proton pump inhibitors, and immune checkpoint inhibitors. Acute interstitial nephritis is likely under-diagnosed, as patients may not present with classic hypersensitivity reactions such as fever, rash, or eosinophilia. Patients with severe bladder outlet obstruction can present with suprapubic pain and a palpable bladder. The use of a portable bladder scanner quickly confirms obstruction by measuring a post-void residual urine volume greater than 100 mL. Renal ultrasonography is sensitive in detecting hydronephrosis, although this characteristic finding of urinary tract obstruction may not be apparent in patients with significant retroperitoneal disease or in the setting of early obstruction.

There has been much controversy as to the optimal solution for fluid resuscitation of the patient with AKI, especially in the setting of sepsis. Mechanisms to maintain renal blood flow through autoregulation are impaired in the setting of AKI. While initial use of fluids to optimize cardiac output is beneficial, prolonged use can lead to harm from fluid overload. A fine balance is often necessary between fluids and vasopressors, as several retrospective studies associate fluid overload with increased mortality [9]. Colloid solutions such as IV albumin and starch have not proven to be more effective than crystalloid solutions and are considerably more expensive [10]. Intravenous starch may cause AKI by inducing osmotic nephrosis of the renal tubules; thus, in general, its use should be avoided in patients with AKI. Albumin and starch leak out of the intravascular compartment within hours after administration, thereby potentially worsening peripheral edema. However, patients with cirrhosis can benefit from IV albumin in the setting of sepsis or large volume paracentesis. We generally prefer using crystalloid solutions such as isotonic saline (0.9% saline) for volume resuscitation. Interestingly, animal studies have demonstrated vasoconstriction of the renal arteries from chloride-containing solutions, which could theoretically worsen renal function. However, balanced (low-chloride) crystalloid fluids such as Plasma-Lyte or lactated Ringer's solution have not consistently demonstrated better outcomes when compared to normal saline [11–13]. Early goal-directed therapy (EGDT) for sepsis (protocol-driven fluid resuscitation, transfusions, inotropes, and vasopressor support) showed promise in a singlecenter RCT [14], but did not decrease mortality or need for renal replacement therapy (RRT) in three subsequent larger RCTs [15–17]. Furthermore, a meta-analysis of individual patient data demonstrated that EGDT had equivalent outcomes but higher hospitalization costs when compared to standard of care [18]. Continuous infusion of norepinephrine (2-12 µg/min) or vasopressin (0.01-0.04 U/min) is generally used if fluid resuscitation alone is unable to maintain a target mean arterial pressure of 65 mmHg. Placement of a Foley catheter should be considered if the patient has signs of bladder outlet obstruction or urinary retention. Emergent placement of a percutaneous nephrostomy (PCN) tube may be necessary if the site of obstruction is above the level of the bladder outlet. The use of nephrotoxic medications and iodinated contrast agents should be avoided, if possible.

RRT is indicated in patients who present with persistent hyperkalemia, fluid overload refractory to diuretics, severe metabolic acidosis, uremia, or marked tumor lysis syndrome (TLS). Early nephrology consultation from the ED expedites dialysis in these patients. Intermittent hemodialysis (IHD) is generally sufficient for volume and metabolic clearance in patients who are hemodynamically stable; however, patients with septic shock or hemodynamic instability may require continuous renal replacement therapy (CRRT) in the intensive care unit. In patients with sepsis and AKI, the use of CRRT has not demonstrated a survival advantage over IHD but can be more effective in minimizing fluid overload [19, 20].

AKI and Conventional Chemotherapy

The most common cause of AKI due to conventional chemotherapy is acute tubular necrosis (ATN). Offending drugs include platinum agents (cisplatin and carboplatin), ifosfamide, and pemetrexed. Cisplatin can cause ATN, thrombotic microangiopathy, and hypomagnesemia. Carboplatin is much less injurious to the kidneys but may still cause ATN and hypomagnesemia. Injury from platinum drugs is generally dose dependent and can lead to chronic kidney disease despite drug discontinuation. ATN from ifosfamide can lead to progressive decline in renal function over several years. Severe injury from ifosfamide manifests as Fanconi's syndrome (hypokalemia, hypophosphatemia, acidosis, and glucosuria), which may persist long after discontinuation of therapy. Pemetrexed, a derivative of methotrexate, can be associated with ATN and diabetes insipidus. Hydration with isotonic fluid is the mainstay to prevent ATN.

High-dose methotrexate (MTX) ranging from 1 to 15 g/ m^2 can precipitate in the renal tubules leading to obstruction,

inflammation, and AKI. Bicarbonate-based intravenous fluids help to clear MTX crystals in the urine by increasing its solubility. Although dialysis lowers serum MTX levels, the effect is temporary due to the high level of protein binding and rebound of serum levels. In patients with impaired renal function, glucarpidase can be administered to metabolize MTX into two inactive compounds that are subsequently cleared by the liver.

Endothelial injury due to gemcitabine, mitomycin C, and cisplatin leads to activation of complement and generation of thrombi within the renal microvasculature leading to thrombotic microangiopathy (TMA). This diagnosis should be suspected in patients with Coombs (–) hemolytic anemia, thrombocytopenia, and AKI. Treatment includes holding the offending drug and possibly the use of complement inhibitors.

AKI and Immunotherapy

Immunotherapy, which is based on regulating the immune system to recognize and eradicate tumor cells, is currently used in the treatment of malignant melanoma, Hodgkin's lymphoma, renal cell and urothelial carcinoma, non-small cell lung cancer, and head/neck cancer. Immune checkpoint inhibitors (ICIs) release inhibition of T-lymphocytes to generate a long-lasting anti-tumor response. Current ICIs target cytotoxic T-lymphocyte-associated antigen 4 (CTLA-4), programmed cell death protein 1 (PD-1), and programmed cell death ligand 1(PD-L1). Renal toxicity occurs in approximately 2% of patients and increases to 5% for patients on dual therapy [21]. This renal injury is usually granulomatous interstitial nephritis, but less commonly presents as lupuslike nephritis, pauci-immune glomerulonephritis, IgA nephropathy, and focal segmental glomerulosclerosis. Onset of kidney disease varies anywhere from 1 to 8 months after initiation of therapy and up to 2 months after the last dose. Nephritis should be considered in any patient treated with ICIs and presenting with AKI, proteinuria, or hematuria. Early administration of steroids should be considered.

AKI and Anti-VEGF Therapy

Angiogenesis inhibition by blocking vascular endothelial growth factor (VEGF) or its receptor inhibits endothelial cell proliferation and vessel formation. Side effects include proteinuria, hypertension, and renal-limited thrombotic microangiopathy in severe cases. Proteinuria due to compromise of the filtration barrier of the nephron ranges from mild to fulminant nephrotic syndrome. Hypertension is mediated by decreased nitrous oxide levels leading to endothelial dysfunction. First-line agents for treatment of hypertension include angiotensin-converting enzyme inhibitors (ACEIs) or angiotensin receptor blockers (ARBs), while calcium channel blockers are suitable alternatives. Centrally acting drugs or diuretics may be added for a goal blood pressure less than 140/90 mm Hg. Patients who develop hypertensive crisis or encephalopathy should discontinue anti-VEGF therapy immediately and admitted to the ICU. Patients who develop TMA in the setting of anti-VEGF therapy will present with de novo hypertension and unexplained renal failure. TMA can resolve with time after withholding further therapy, although some patients will benefit from anti-complement therapy such as eculizumab.

Chronic Kidney Disease in Cancer Patients

The prevalence of CKD ranges from 12% to 53% at the time of cancer diagnosis [22, 23]. Older patients with cancer have a relatively higher incidence of hypertension and diabetes, which are the two most common risk factors for CKD. Cancer therapy is also associated with progressive CKD. A retrospective study of patients with solid tumors demonstrated a reduction in eGFR of 13 mL/min/1.73 m² after 2 years, and 17.7% of patients with CKD stage II advanced to stage III or IV [23]. The incidence of CKD is even higher in patients with kidney cancer and bladder cancer (28.7% and 46%, respectively) [24, 25].

Glomerular filtration rate (GFR) as measured by inulin or iohexol clearance is the gold standard for measurement of kidney function but is difficult to perform in the clinical setting. Therefore, the estimated GFR (eGFR) as calculated by formulas using SCr, age, gender, and race is commonly reported by most hospital laboratories. Estimates of GFR as measured by the Modification of Diet in Renal Disease (MDRD) [26] and Chronic Kidney Disease Epidemiology Collaboration (CKD-EPI) [27] equations are felt to be most accurate, and have largely supplanted the older Cockcroft Gault (CG) equation [28]. However, most pharmacokinetic studies of drugs were based on the CG equation. It is important to remember several caveats concerning estimating equations for GFR. Ninety percent of patients will have a true GFR within 30% of their eGFR. The equations are less valid in patients with extremes in body mass. Additionally, they tend to underestimate GFR in patients with normal or near-normal SCr levels. Most importantly, these equations are not reliable estimates when the SCr is not in steady state as seen in AKI. It should be assumed that any patient that has oliguric AKI unresponsive to hydration has a GFR <10 ml/ min regardless of the SCr level.

Most estimates of glomerular filtration rate (eGFR) are based on SCr levels that are affected by many different factors. Elevation in SCr is a relatively late marker of CKD, and only manifests after a 50% decline in glomerular filtration rate. Reduced creatinine production due to acute illness and sarcopenia, as well as creatinine dilution during volume overload, can further complicate the estimation of renal function in patients with cancer.

Multiple Myeloma and AKI

Multiple myeloma is a clonal malignancy of plasma cells that results in the overproduction of immunoglobulins, their fragments, or free light chains that circulate in the blood (paraproteins). These paraproteins cause injury by depositing in peripheral organs such as the heart, kidney, and liver. AKI from paraprotein deposition in the kidneys is often the initial presentation of patients with multiple myeloma. Cast formation in the distal tubule occurs when paraproteins filter through the glomeruli and bind to Tamm-Horsfall mucoprotein causing obstruction, tubular injury, and inflammation. Amyloid light chain amyloidosis (AL amyloidosis) develops when paraproteins undergo conformational changes and deposit as microscopic fibrils in the glomeruli and vessels. Light chains or heavy chains are deposited within the glomerular and tubular basement membranes, leading to monoclonal immunoglobulin deposition disease (MIDD).

The clinical presentation of AKI in patients with multiple myeloma varies from asymptomatic proteinuria to nephrotic syndrome and rapidly progressive renal failure. AKI is diagnosed in more than half of all patients with multiple myeloma at initial presentation, of which 10% require dialysis. Multiple myeloma should always be part of the differential diagnosis in elderly patients with unexplained acute or chronic kidney disease. Initial workup for multiple myeloma consists of serum and urine protein electrophoresis, immunofixation electrophoresis, and serum-free light chain assays to detect a monoclonal protein. Monoclonal proteins in the urine (Bence-Jones proteins) are not detected by routine qualitative dipstick urinalysis, which detects mainly albuminuria. However, paraprotein deposits from MIDD and AL amyloidosis can damage the filtration barrier of the glomerulus, leading to significant albuminuria detectable on dipstick urinalysis. In contrast, myeloma cast nephropathy has minimal glomerular involvement and typically presents with only mild albuminuria. Other clinical manifestations from light chain amyloid deposits include restrictive cardiomyopathy, hepatomegaly, carpal tunnel syndrome, and orthostatic hypotension. Definitive diagnosis is confirmed by a renal biopsy revealing characteristic casts, light chains, or amyloid deposits.

Early aggressive treatment of patients presenting with multiple myeloma and renal disease helps stabilize or improve kidney function. Initial hydration consists of normal saline infusion, with a urine output goal of 2.5–3.0 L a day, which helps prevent precipitation of casts within the distal tubule. Steroids are often used initially to decrease the production of paraproteins and alleviate end-organ damage. Aminoglycosides, IV contrast agents, diuretics, and nonsteroidal anti-inflammatory drugs (NSAIDs) exacerbate renal injury and should be avoided. Hypercalcemia commonly occurs in patients with multiple myeloma and aggravates acute kidney injury. If hypercalcemia does not resolve with the use of hydration and calcitonin, therapy with a bisphosphonate should be considered (e.g., 3-4 mg of zoledronic acid diluted in 100 mL of normal saline administered in an IV infusion for at least 15 min). Plasmapheresis to remove circulating paraproteins has not been shown to improve clinical outcomes. Similarly, the use of high-cutoff filters in hemodialysis, which is much more effective in removing paraproteins than plasmapheresis, does not definitely improve outcomes when used concurrently with effective chemotherapy [29, 30].

Hematopoietic Stem Cell Transplantation (HSCT) and AKI

AKI is a frequent complication of HSCT and has an incidence between 10% and 73% depending on the type of transplant and conditioning regimen [31]. Risk factors include older age, female gender, allogeneic transplant, myeloablative regimen, graft versus host disease (GVHD), hypertension, diabetes, and preexisting CKD. Common causes of AKI include nephrotoxic drugs (e.g., calcineurin inhibitors, foscarnet, amphotericin), sepsis, volume depletion, hepatic sinusoidal obstruction syndrome (HSOS), thrombotic microangiopathy (TMA), and viral nephritis (e.g., BK virus, adenovirus).

Workup in the ED should include a thorough search for nephrotoxic medications and assessment of volume status. A urinalysis, urine protein/creatinine (UPC) ratio, urine sodium, and urine creatinine help determine the etiology of AKI. A low urine sodium level <20 meq/L or fractional excretion of sodium (FENa) <1% suggests volume depletion. Patients with GVHD of the GI tract can present with prerenal azotemia due to vomiting and diarrhea.

HSOS occurs due to endothelial damage of the liver sinusoids from the conditioning regimen given prior to transplant. Patients with HSOS generally present early after HSCT with jaundice, right upper quadrant pain, ascites, edema, and hepatorenal syndrome. Other causes of cholestasis such as GVHD, sepsis, fungal infection, and TPN should be excluded. Doppler imaging of the portal vein in the setting of HSOS reveals reversal of flow. Initial management includes sodium restriction and diuretics, while RRT will be necessary for severe cases.

Viral infections of the genitourinary tract (e.g., adenovirus, BK virus) cause gross hematuria and bladder outlet obstruction from clots. A three-way urethral catheter should be inserted for continuous bladder irrigation, and urologic consultation should be obtained. Measurement of serum and urine BK and adenovirus levels by PCR should be sent.

Paraneoplastic glomerular disease such as membranous nephropathy or minimal change disease can occur months to years after HSCT while tapering immunosuppression. This diagnosis should be suspected in patients with a history of HSCT that present to the hospital with de novo nephrotic range proteinuria and edema. Symptoms are initially managed with diuretics, but renal biopsy is required for definitive diagnosis.

Electrolyte Abnormalities

Tumor Lysis Syndrome (TLS)

TLS is a common life-threatening emergency in patients with cancer presenting to the ED. Tumor cells rapidly release potassium, phosphorus, and uric acid into the extracellular space and overwhelm the excretory capacity of the kidneys. Hyperkalemia predisposes patients to cardiac dysrhythmias and sudden death. Hyperphosphatemia and secondary hypocalcemia may lead to AKI, muscular irritability, cardiac dysrhythmias, and metastatic calcification. Uric acid precipitating as crystals in the renal tubules causes obstruction, vasoconstriction, and inflammation. Patients with rapidly proliferating chemosensitive hematologic malignancies are at the greatest risk for TLS. Risk factors for TLS include a white blood cell count greater than 50,000/µL, elevated lactate dehydrogenase level, bulky disease, marrow or organ infiltration, advanced age, and chronic kidney disease. Cases of TLS in patients with a solid tumor undergoing chemotherapy and/or radiation therapy have been reported; however, patients with non-Hodgkin's lymphoma or acute leukemia are at greatest risk.

The identification of TLS is generally straightforward in patients who present with marked derangements in electrolyte levels. However, patients with impaired renal function secondary to an effective prerenal state, such as volume depletion or hypotension, can also develop hyperkalemia, hyperphosphatemia, and hyperuricemia. Unlike patients with TLS, patients with prerenal azotemia will rapid normalize electrolyte levels and renal function with hydration and optimization of blood pressure.

Intravenous hydration to maintain adequate urine output and xanthine oxide inhibitors (allopurinol or febuxostat) to prevent the formation of uric acid are the mainstays of preventing TLS. Infusion of isotonic saline should be instituted 24 h prior to chemotherapy at 100 mL/m²/h and titrated accordingly to maintain a urine output of at least 2.5 L a day. Conservative fluid management strategies are necessary in patients with underlying congestive heart failure. High-risk patients or those who have elevated uric acid levels prior to chemotherapy should receive rasburicase (0.2 mg/kg [IV] daily for up to 5 days). Rasburicase converts uric acid to allantoin, which is 5–10 times more soluble in the urine. Fixed dosing of rasburicase (3 or 6 mg) up to 2 doses has been shown to be just as effective as weight-based dosing and leads to significant cost savings [32]. Rasburicase causes production of hydrogen peroxide that can cause hemolytic anemia in patients with glucose-6-phosphate dehydrogenase (G6PD) deficiency. Patients at risk should have G6PD levels checked prior to receiving rasburicase.

Management of established TLS also consists of judicious hydration as well as rasburicase for hyperuricemia. Alkalization of the urine retards the formation of uric acid crystals but increases the risk of calcium phosphate crystal deposition. Therefore, routine urine alkalization in patients with TLS is no longer recommended. Nephrology consultation should be sought for patients presenting to the EC with TLS who have peaked T waves on electrocardiogram (EKG), dysrhythmias, or oliguria. Angiotensin-converting enzyme inhibitors, angiotensin receptor blockers, and NSAIDs should be avoided in patients with TLS because they worsen hyperkalemia and AKI. Although some patients deemed high risk of developing severe TLS have been preemptively started on CRRT prior to chemotherapy, this is not standard practice. Pseudohyperkalemia occurs in the setting of extreme leukocytosis, due to spurious elevations in serum potassium levels resulting from ex vivo lysis of white blood cells. The diagnosis should be considered in patients without other signs of TLS and the absence of EKG changes suggestive of hyperkalemia. If a patient is suspected of having pseudohyperkalemia, potassium level should be measured on a heparinized plasma sample placed on ice or on the whole blood by a point-of-care analyzer.

Hyponatremia

Hyponatremia (serum sodium levels less than 135 mEq/L) has been reported in 14–23% of patients with cancer on admission to the hospital. Increased hyponatremia severity is a negative prognostic factor for survival in patients with nonsmall cell lung cancer, renal cell carcinoma, gastric cancer, and non-Hodgkin's lymphoma. The two most frequent causes of hyponatremia in patients with cancer are hypovolemia and syndrome of inappropriate antidiuretic hormone secretion (SIADH). Common etiologies of SIADH include malignancy (e.g., the lung, gastrointestinal, central nervous system), pneumonia, drugs (e.g., antidepressants, haloperidol, carbamazepine, cyclophosphamide, platinum compounds, vinca alkaloids), nausea, and pain. Other considerations when assessing patients with hyponatremia in the setting of cancer include renal salt wasting secondary to chemotherapy, "tea and toast syndrome" resulting from malnutrition, water intoxication, and adrenal insufficiency secondary to adrenal metastases or steroid withdrawal.

Symptoms in patients with hyponatremia may be absent, mild (confusion, dizziness, nausea, and lethargy), or severe (seizures, coma, and death). The occurrence of symptoms depends primarily on the rate of decline in the serum sodium level as opposed to the absolute level. Adaptation of the brain to hyponatremia occurs gradually by the excretion of osmolytes from cells to prevent cerebral edema. Cerebral edema with eventual brain stem herniation is a risk if the rate of decline in serum sodium level outpaces the excretion of osmolytes. Immediate treatment is required to raise the serum sodium level until the patient is asymptomatic. If the decline in serum sodium level is more gradual, the patient may be asymptomatic or have only mild symptoms. Rapid correction of hyponatremia is not indicated in this situation.

Initial workup for hyponatremia should include a physical examination to assess the patient's volume status, chemistry profile, plasma osmolality, urine electrolyte levels, and urine osmolality. Patients with volume depletion generally have urine sodium levels less than 20 mEq/L and concentrated urine (urine osmolality greater than plasma osmolality). Patients with hypervolemia (those with heart failure, cirrhosis, third spacing caused by peritoneal or liver metastases, hypoalbuminemia, or inferior vena cava compression or obstruction) have signs of fluid overload upon physical examination (e.g., edema, ascites, effusions) but are in an effectively prerenal state. Therefore, they also will have urine sodium levels less than 20 mEq/L and concentrated urine. Patients with SIADH have urine sodium levels greater than 40 mEq/L and inappropriately dilute urine (urine osmolality less than plasma osmolality). Patients with "tea and toast syndrome" have serum sodium levels less than 20 mEq/L and appropriately dilute urine (urine osmolality less than plasma osmolality). Urine sodium levels are variable in patients with water intoxication, but urine osmolality is appropriately dilute, typically less than 150 mOsm/kg.

Urgent treatment is not indicated for hyponatremic patients who are asymptomatic or have only mild symptoms. Rapid correction of hyponatremia in these patients will increase the risk of osmotic demyelination syndrome. Patients with volume depletion should receive isotonic fluids such as normal saline. Otherwise, total fluid intake should be restricted to less than 1 L daily. Loop diuretics should be considered in patients with hypervolemia. The treatment of patients with SIADH consists of fluid restriction, salt tablets (initially, 1 g three times daily) and possibly, loop diuretics. The introduction of vasopressin receptor antagonists has revolutionized treatment of hyponatremia in patients with hypervolemia or SIADH. These drugs block the effect of antidiuretic hormone on the collecting ducts of the kidney, thereby preventing water reabsorption and stimulating water diuresis. Currently, two drugs within this class are available, oral tolvaptan (initially 7.5–15.0 mg daily) and IV conivaptan (20-mg loading dose with 20 mg administered over the ensuing 24 h). Serial sodium levels should be monitored every 4–6 h with a goal correction rate of less than 8 mEq/L in 24 h.

Patients with severe symptoms require more urgent intervention with hypertonic saline and close neurologic monitoring. Treatment consists of infusion of 3% saline at a rate of 0.8 mL/kg/h and initial monitoring of serum sodium levels at least every 2–4 h. Alternatively, 3% saline can be administered as 100 ml bolus over 10 min with repeat dosing as needed. The infusion is continued until the sodium level is greater than 120 mEq/L, symptoms have resolved, or the rate of sodium level correction has exceeded 8 mEq within 24 h. Rates of correction in excess of 10–12 mEq per 24 h causes osmotic demyelination syndrome, which results in altered mental status, quadriparesis, quadriplegia, pseudobulbar palsy, coma, or death. Therefore, frequent neurologic assessments and titration of the 3% saline infusion to prevent overcorrection of hyponatremia are necessary.

Hyperkalemia

Hyperkalemia is a particular concern in patients presenting to the ED, as it can be life threatening. Common causes of hyperkalemia in patients with cancer include AKI, TLS, the use of certain drugs (e.g., calcineurin inhibitors, NSAIDs, angiotensin-converting enzyme inhibitors, trimethoprimsulfamethoxazole, potassium-sparing diuretics) and metabolic acidosis. In most patients, the etiology is multifactorial.

Most patients with hyperkalemia are clinically asymptomatic unless their potassium levels are very high. Cardiac dysrhythmia is the most concerning manifestation. Electrocardiographic changes progress from peaked T waves, flattened P waves, and widened QRS complexes to eventual sine waves. It should be noted that up to 50% of patients with significant hyperkalemia may not manifest EKG abnormalities. Skeletal muscle weakness should also be present in the setting of severe hyperkalemia. Stabilization of the myocardial membrane with IV administration of calcium gluconate (10 mL of 10% IV) or calcium chloride (2 g over 5 min) to counter the effects of hyperkalemia is imperative. A repeat infusion of calcium may be necessary if EKG changes persist. Temporizing measures that shift potassium to the intracellular space involve administration of regular insulin (10 U intravenous), glucose (50 mL of 50% dextrose intravenous), and nebulized albuterol (20 mg in 4 mL). Sodium bicarbonate administration is helpful in patients with concurrent metabolic acidosis by shifting potassium to

the intracellular space as the acidosis is corrected. If severe hyperkalemia persists despite correction of its underlying cause, an urgent nephrology consultation should be sought as dialysis might be necessary. For patients with milder hyperkalemia, monitoring serial potassium levels after discontinuation of the causative drug is sufficient. Loop diuretics can be administered to patients with adequate renal function to enhance potassium excretion in the urine. Sodium polystyrene sulfonate (15-30 g oral) is commonly given to facilitate potassium elimination from the gut; however, its clinical effectiveness is unproven in clinical studies, and its use is rarely associated with intestinal necrosis [33]. Newer potassium-binding agents, patiromer (K+/Ca2+ exchange) and sodium zirconium cyclosilicate (K+/Na+ and H+ exchange), are now clinically available. These newer agents have a more predictable onset of action with less incidence of diarrhea.

Hypercalcemia

Hypercalcemia occurs in 20-30% of all malignancies and is the most common paraneoplastic syndrome. The majority (~80%) of cases are mediated by production of parathyroid hormone-related peptide (PTHr-P) as seen in squamous cell carcinoma, breast cancer, kidney cancer, prostate cancer, and bladder cancer. Local osteolysis due to cytokine release comprises the remaining cases. Lymphoma may cause upregulation of 1,25-hydroxy vitamin D causing hypercalcemia. Rarely, ectopic PTH production occurs in certain tumors. Hypercalcemia causes AKI by vasoconstriction of the afferent arteriole and hypovolemia from polyuria. Intravenous saline should be started to maintain urine output greater than 100 mL/h. Calcitonin begins to decrease serum calcium levels within 15 min of IM or SC administration and has a peak effect within 4 h. Many patients will require a more durable response with osteoclast inhibitors such as pamidronate or zoledronic acid. Denosumab, an inhibitor of receptor activator of nuclear factor-kappaB (RANK) ligand, also provides a long-lasting response to normalize calcium levels. However, patients can develop profound hypocalcemia several weeks later if they have concurrent vitamin D deficiency.

Key Practice Points

- Patients with renal cell carcinoma, multiple myeloma, leukemia, and lymphoma are at highest risk of developing AKI.
- The KDIGO criteria provide a standardized definition of AKI based on increases in serum creatinine level relative to baseline and have prognostic value in the care of patients with cancer.

- Immune checkpoint inhibitors cause AKI by interstitial nephritis, glomerulonephritis, or vasculitis.
- Renal toxicity from anti-VEGF therapy presents with a spectrum of disease from mild proteinuria to malignant hypertension.
- More than half of all patients with multiple myeloma will initially present with some degree of renal injury, and AKI improves with immediate treatment of the underlying myeloma.
- Treatment of TLS includes aggressive IV hydration, rasburicase for hyperuricemia, and, possibly, dialysis for AKI.
- Vasopressin receptor antagonist drugs have revolutionized the treatment of hyponatremia associated with hypervolemia or SIADH, but administration of 3% saline is still required for patients with hyponatremia and severe symptoms (seizures or coma).
- Hyperkalemia may not manifest clinically until potassium levels are severely elevated, and emergent treatment of it includes IV calcium to stabilize the myocardial membrane, IV insulin with glucose, inhaled beta-agonists, and dialysis in refractory cases.

References

- Christiansen CF, Johansen MB, Langeberg WJ, Fryzek JP, Sorensen HT. Incidence of acute kidney injury in cancer patients: a Danish population-based cohort study. Eur J Intern Med. 2011;22(4):399–406.
- Kitchlu A, McArthur E, Amir E, Booth CM, Sutradhar R, Majeed H, et al. Acute kidney injury in patients receiving systemic treatment for cancer: a population-based cohort study. J Natl Cancer Inst. 2019;111(7):727–36.
- Lahoti A, Nates JL, Wakefield CD, Price KJ, Salahudeen AK. Costs and outcomes of acute kidney injury in critically ill patients with cancer. J Support Oncol. 2011;9(4):149–55.
- 4. Bellomo R, Ronco C, Kellum JA, Mehta RL, Palevsky P, Acute Dialysis Quality Initiative workgroup. Acute renal failure - definition, outcome measures, animal models, fluid therapy and information technology needs: the second international consensus conference of the acute dialysis quality initiative (ADQI) group. Crit Care. 2004;8(4):R204–12.
- Uchino S, Bellomo R, Goldsmith D, Bates S, Ronco C. An assessment of the RIFLE criteria for acute renal failure in hospitalized patients. Crit Care Med. 2006;34(7):1913–7.
- Ali T, Khan I, Simpson W, Prescott G, Townend J, Smith W, et al. Incidence and outcomes in acute kidney injury: a comprehensive population-based study. J Am Soc Nephrol. 2007;18(4):1292–8.
- Lahoti A, Kantarjian H, Salahudeen AK, Ravandi F, Cortes JE, Faderl S, et al. Predictors and outcome of acute kidney injury in patients with acute myelogenous leukemia or high-risk myelodysplastic syndrome. Cancer. 2010;116(17):4063–8.
- Kaddourah A, Basu RK, Bagshaw SM, Goldstein SL, Investigators A. Epidemiology of acute kidney injury in critically ill children and young adults. N Engl J Med. 2017;376(1):11–20.
- 9. Bouchard J, Soroko SB, Chertow GM, Himmelfarb J, Ikizler TA, Paganini EP, et al. Fluid accumulation, survival and recovery of

kidney function in critically ill patients with acute kidney injury. Kidney Int. 2009;76(4):422–7.

- Lewis SR, Pritchard MW, Evans DJ, Butler AR, Alderson P, Smith AF, et al. Colloids versus crystalloids for fluid resuscitation in critically ill people. Cochrane Database Syst Rev. 2018;(8):CD000567.
- 11. Young P, Bailey M, Beasley R, Henderson S, Mackle D, McArthur C, et al. Effect of a buffered crystalloid solution vs saline on acute kidney injury among patients in the intensive care unit: the SPLIT randomized clinical trial. JAMA. 2015;314(16):1701–10.
- Semler MW, Self WH, Wanderer JP, Ehrenfeld JM, Wang L, Byrne DW, et al. Balanced crystalloids versus saline in critically ill adults. N Engl J Med. 2018;378(9):829–39.
- Self WH, Semler MW, Wanderer JP, Wang L, Byrne DW, Collins SP, et al. Balanced crystalloids versus saline in noncritically ill adults. N Engl J Med. 2018;378(9):819–28.
- Rivers E, Nguyen B, Havstad S, Ressler J, Muzzin A, Knoblich B, et al. Early goal-directed therapy in the treatment of severe sepsis and septic shock. N Engl J Med. 2001;345(19):1368–77.
- Mouncey PR, Osborn TM, Power GS, Harrison DA, Sadique MZ, Grieve RD, et al. Trial of early, goal-directed resuscitation for septic shock. N Engl J Med. 2015;372(14):1301–11.
- Pro CI, Yealy DM, Kellum JA, Huang DT, Barnato AE, Weissfeld LA, et al. A randomized trial of protocol-based care for early septic shock. N Engl J Med. 2014;370(18):1683–93.
- ARISE Investigators, ANZICS Clinical Trials Group, Peake SL, Delaney A, Bailey M, Bellomo R, et al. Goal-directed resuscitation for patients with early septic shock. N Engl J Med. 2014;371(16):1496–506.
- Investigators P, Rowan KM, Angus DC, Bailey M, Barnato AE, Bellomo R, et al. Early, goal-directed therapy for septic shock - a patient-level meta-analysis. N Engl J Med. 2017;376(23):2223–34.
- Nash DM, Przech S, Wald R, O'Reilly D. Systematic review and meta-analysis of renal replacement therapy modalities for acute kidney injury in the intensive care unit. J Crit Care. 2017;41:138–44.
- Rabindranath K, Adams J, Macleod AM, Muirhead N. Intermittent versus continuous renal replacement therapy for acute renal failure in adults. Cochrane Database Syst Rev. 2007;(3):CD003773.
- Cortazar FB, Marrone KA, Troxell ML, Ralto KM, Hoenig MP, Brahmer JR, et al. Clinicopathological features of acute kidney injury associated with immune checkpoint inhibitors. Kidney Int. 2016;90(3):638–47.

- Launay-Vacher V. Epidemiology of chronic kidney disease in cancer patients: lessons from the IRMA study group. Semin Nephrol. 2010;30(6):548–56.
- Launay-Vacher V, Oudard S, Janus N, Gligorov J, Pourrat X, Rixe O, et al. Prevalence of renal insufficiency in cancer patients and implications for anticancer drug management: the renal insufficiency and anticancer medications (IRMA) study. Cancer. 2007;110(6):1376–84.
- 24. Lane BR, Demirjian S, Derweesh IH, Takagi T, Zhang Z, Velet L, et al. Survival and functional stability in chronic kidney disease due to surgical removal of nephrons: importance of the new baseline glomerular filtration rate. Eur Urol. 2015;68(6):996–1003.
- Eisenberg MS, Thompson RH, Frank I, Kim SP, Cotter KJ, Tollefson MK, et al. Long-term renal function outcomes after radical cystectomy. J Urol. 2014;191(3):619–25.
- 26. Levey AS, Bosch JP, Lewis JB, Greene T, Rogers N, Roth D. A more accurate method to estimate glomerular filtration rate from serum creatinine: a new prediction equation. Modification of diet in renal disease study group. Ann Intern Med. 1999;130(6):461–70.
- 27. Levey AS, Stevens LA, Schmid CH, Zhang YL, Castro AF 3rd, Feldman HI, et al. A new equation to estimate glomerular filtration rate. Ann Intern Med. 2009;150(9):604–12.
- Cockcroft DW, Gault MH. Prediction of creatinine clearance from serum creatinine. Nephron. 1976;16(1):31–41.
- Bridoux F, Carron PL, Pegourie B, Alamartine E, Augeul-Meunier K, Karras A, et al. Effect of high-cutoff hemodialysis vs conventional hemodialysis on hemodialysis independence among patients with myeloma cast nephropathy: a randomized clinical trial. JAMA. 2017;318(21):2099–110.
- 30. Hutchison CA, Cockwell P, Moroz V, Bradwell AR, Fifer L, Gillmore JD, et al. High cutoff versus high-flux haemodialysis for myeloma cast nephropathy in patients receiving bortezomib-based chemotherapy (EuLITE): a phase 2 randomised controlled trial. Lancet Haematol. 2019;6(4):e217–e28.
- Renaghan AD, Jaimes EA, Malyszko J, Perazella MA, Sprangers B, Rosner MH. Acute kidney injury and CKD associated with hematopoietic stem cell transplantation. Clin J Am Soc Nephrol. 2020;15(2):289–97.
- 32. McBride A, Lathon SC, Boehmer L, Augustin KM, Butler SK, Westervelt P. Comparative evaluation of single fixed dosing and weight-based dosing of rasburicase for tumor lysis syndrome. Pharmacotherapy. 2013;33(3):295–303.
- Sterns RH, Rojas M, Bernstein P, Chennupati S. Ion-exchange resins for the treatment of hyperkalemia: are they safe and effective? J Am Soc Nephrol. 2010;21(5):733–5.



Shawn Dason, Cheryl T. Lee, and Ahmad Shabsigh

Check for updates

31

Case Study

A 72-year-old male smoker presented with lower abdominal pain and an inability to urinate. He endorsed several months of gross hematuria, fatigue, and worsening urinary symptoms. On presentation, he was tachycardic, hypertensive, and had acute renal insufficiency with a creatinine of 4 mg/ dL. There were no indications for dialysis. The on-call urologist was contacted early (Fig. 31.1) and recommended catheterization with a 24-Fr 3-way Folev catheter as they made their way to assess the patient. Catheterization and manual irrigation resulted in evacuation of a significant amount of blood clot along with resolution of pain, tachycardia, and hypertension. Despite 45 min of manual irrigation, significant hematuria and clot continued. A noncontrast CT scan revealed bilateral hydroureteronephrosis with a transition point at the level of the bladder. The bladder was distended with hyperdense material concerning for a large bladder mass or blood clot. The patient was admitted by the urology service for continuous bladder irrigation. The next morning, the patient's hematuria continued and creatinine remained elevated. He was taken to the operating room for an endoscopic assessment and ureteral stent placement. Cystoscopy revealed a large hemorrhagic bladder mass that was completely resected for diagnosis and management of the bleeding. The ureteral orifices were not found and ureteral stents could not be placed. The patient was taken to the interventional radiology suite for bilateral nephrostomy tube placement. Creatinine subsequently normalized although there was a period of post-obstructive diuresis. Pathology from the transurethral resection of his bladder tumor revealed urothelial carcinoma with involvement of the muscularis propria. After his hematuria, elevated creatinine and diuresis normalized, the patient was discharged. He had multidisciplinary consultation for his bladder cancer as an outpatient.

Following negative staging scans, he received neoadjuvant chemotherapy with four cycles of gemcitabine and cisplatin with his nephrostomy tubes in situ. He then had a radical cystectomy and ileal conduit.

Hematuria

Hematuria is a common urologic issue assessed in the emergency department (ED). Hematuria can be divided into gross (or visible) hematuria and microhematuria. While the rate of a cancer diagnosis is 1–3% with microhematuria [1, 2], this rises to 25% with gross hematuria [3]. Discolored urine [4], vaginal bleeding, or rectal bleeding may sometimes mimic hematuria and should be considered. True hematuria can usually be determined by history or examination of the urine. If the diagnosis of hematuria is in question, microscopy can be performed to confirm the presence of visualized red blood cells in the urine.

Microhematuria

Microhematuria refers to red blood cells in the urine visualized on microscopy but not with the naked eye. A threshold of three or more red blood cells per high-powered field is traditionally used to define microhematuria [1]. Although some definitions accept chemical urinalysis (dipstick) findings of blood as being consistent with microhematuria, it is generally thought that microscopy is required due to the potential for dipstick false positives (e.g., myoglobinuria) [5]. Microhematuria in the ED setting can often be an incidental finding as it is present in 6.5% of the general population [1]. Numerous nonpathologic causes exist for microhematuria including instrumentation, dehydration, exercise, menses, and other intraabdominal pathology. Nonetheless, the differential diagnosis for microhematuria includes numerous pathologies and its presence should warrant consideration of a urologic etiology to the presenting

© Springer Nature Switzerland AG 2021

S. Dason (⊠) · C. T. Lee · A. Shabsigh Department of Urology, The Ohio State University, Columbus, OH, USA e-mail: Shawn.Dason@osumc.edu

K. H. Todd et al. (eds.), Oncologic Emergency Medicine, https://doi.org/10.1007/978-3-030-67123-5_31

Fig. 31.1 A management algorithm for gross hematuria presenting to the ED. Please refer to the text for additional details



complaint. Microhematuria associated with proteinuria, casts, hypertension, or renal dysfunction should also prompt a nephrologic workup [1]. While microhematuria alone does not generally require any acute management, microhematuria outside of the context of a clearly documented infection warrants outpatient follow-up with the patient's primary care physician and/or urologist. This can be especially important as the rate of cancer diagnosis with microhematuria is 1-3% [2, 6]. Risk factors for malignancy are numerous but include symptomatic microhematuria, smoking status, degree of microhematuria, exposure to pelvic radiation or chemicals, and age [1, 2].

Gross Hematuria

Gross hematuria is visibly apparent to the patient and on examination of the urine. Nonetheless, gross hematuria may be intermittent, and a normal urinalysis, after symptoms have resolved, does not obviate a subsequent outpatient workup. Gross hematuria is likely urologic in origin with a cause identified in 50% of patients and a malignant etiology identified in 25% [3]. Although rare, a nephrologic cause should be considered in patients with red blood cell casts on microscopy, proteinuria, hypertension, and renal dysfunction [1].

The first step in assessing gross hematuria is to identify and manage any associated hemodynamic compromise because hemorrhagic shock can occasionally occur (see Fig. 31.1). Consideration should be given to an alternate etiology of hypotension as this is infrequent. Aggressive resuscitation should be undertaken in parallel with urgent urology consultation in the unstable patient as certain scenarios may necessitate immediate transfer straight to the operating room or interventional radiology suite.

The stable patient with gross hematuria should be assessed for urinary retention (see Fig. 31.1). Blood in the urinary tract will have a tendency to clot and obstruct the lower urinary tract or an indwelling urinary catheter. In this scenario, patients may endorse an inability to urinate, urinary urgency, a reduction in urine output, overflow incontinence, suprapubic pain, abdominal distension, or edema. Signs include generalized discomfort, a palpable painful bladder, visible voided blood clots or a clotted urinary catheter, edema, hypertension, and tachycardia.

The patient with suspected clot urinary retention should be catheterized emergently as long as there are no indications for urology consultation prior to catheter placement (see Fig. 31.1). Generally those with recent urinary tract surgery, trauma, or failed catheterization attempts should be catheterized only by a urologist due to the potential for further trauma to the urinary tract [7]. Patients should be catheterized in a private setting where vital signs can be measured, as a vagal response can sometimes result following bladder

decompression [8]. In the setting of suspected clot retention. the initial catheter placed should be a 3-way catheter that has a large-caliber (22 Fr or 24 Fr). A 3-way catheter should be placed to facilitate continuous bladder irrigation with saline should this be required following bladder drainage and manual irrigation. A larger catheter (22 or 24 Fr) will allow for significantly improved evacuation of blood clots (initially with hand irrigation of the Foley) than the conventional first-line urinary catheter used in other situations (16 Fr) [9]. Special "hematuria catheters" are commercially available and may be stocked in the urology supply area of the operating room. If available, these catheters are ideal for this clinical situation, as they are large-bore 22- or 24-Fr 3-way catheters with a large opening at the tip for clot evacuation and are stiffer to prevent catheter collapse from aspiration. Initial placement of an appropriate catheter will permit efficient clearance of obstructive clots and minimize subsequent catheter changes. Lubricating the catheter well and use of 2% lidocaine gel intraurethrally can minimize discomfort in the placement of a larger catheter. If the patient is not in painful acute urinary retention, waiting for 10-20 min after injection of the gel prior to catheter placement is advised to allow for the lidocaine to take effect. A urologist should be involved following any concerns about catheterization failure such as ongoing symptoms, inability to advance the catheter to its hub, lack of drainage on catheter insertion, inability to irrigate a catheter, discomfort or excess resistance with irrigation, or Foley balloon inflation.

In the event that the patient presents with urinary retention and an obstructed catheter, the catheter should first be flushed with a cone-tipped (Toomey) syringe and sterile saline to relieve the obstruction. Ensure that a cone-tipped syringe is being used since a Luer-lock syringe does not have a tip diameter large enough to aspirate clots. The largest port on the catheter (center port in a 3-way catheter) should be used for irrigation and aspiration. After the bladder has started draining and discomfort is resolving, clot irrigation should be performed. If the catheter blocks and cannot be unobstructed by flushing, it may require replacement. Before removing the obstructed catheter, any anticipated challenges with replacing it should be carefully considered. Especially in the setting of recent urinary tract surgery, a urologist should generally be contacted before removal and/or replacement of the obstructed catheter.

After catheter placement and initial bladder decompression, further manual irrigation should be performed to evacuate the bladder of clots. If the catheter is draining, manual irrigation may be delayed to ensure that the bladder drains as much as possible, vagal response has not occurred, basic labs are sent, and any necessary imaging requests or consultations are in process. The catheter should be well lubricated prior to manual irrigation, as subtle motions during catheter manipulation can be uncomfortable. Manual irrigation is performed through the middle (largest caliber) port of a 3-way catheter or the larger caliber port of a 2-way catheter. For both a 2-way and 3-way catheters, the Luer-lock port is used to inflate the balloon only. A 60 ml sterile syringe is filled with sterile saline and instilled slowly. In the case of a low-capacity bladder or a bladder filled with clot, the patient may not tolerate a full 60 cc and overfilling is suggested by increased resistance to fluid instillation or excessive patient discomfort. Overfilling the bladder can cause bladder trauma. After instillation, the syringe should be gently aspirated to draw back clot. Thick dense clot can sometimes require some force to aspirate, although care should be taken to ensure that one is not aspirating the bladder mucosa, which may be in contact with the catheter tip. If a regular (i.e., 16 Fr) catheter is in place and aspiration is not proceeding smoothly, a larger catheter should generally be placed if no contraindications to catheter exchange exist. If 60 ml of instillation is tolerated but limited clot returns on aspiration, a number of maneuvers can be attempted to completely irrigate the bladder. First, an additional 60 cc can be instilled into the bladder such that irrigation can now take place with 120 cc of fluid in the bladder. This additionally distends the bladder, reducing contact of bladder mucosa and the catheter eyelet. Rotating the catheter during irrigation, withdrawing or advancing the catheter, rapidly alternating a low-volume instillation aspiration, and deflating the balloon can also be helpful maneuvers to maximize clot irrigation. Clot irrigation can sometimes require 30-60 min and a reasonable attempt should be made to clear the bladder of clot prior to moving onwards.

Once the bladder has been completely evacuated of clots, consideration is given to continuous bladder irrigation (CBI) (see Fig. 31.1). CBI should not be used unless manual evacuation has ensured that large clots have been evacuated from the bladder. This is because these clots are unlikely to clear with CBI alone and may obstruct the catheter during passage. An obstructed catheter outflow with unattended continuous inflow in the setting of CBI can result in bladder trauma and recurrent symptoms of retention. CBI does not replace manual irrigation but can be useful to ensure that ongoing bleeding is efficiently evacuated from the bladder such that large clots will not form. Large bags of saline are connected to a cone-tipped tubing that will be attached to the small cone-tipped port of a 3-way catheter. Generally Y-tubing should be used to facilitate hanging two bags at once such that the need to exchange bags can be minimized. Only one bag should be unclamped and actively flowing at a time. The rate of continuous bladder irrigation can be varied by adjusting the tightness of a valve that will be present on CBI Y-tubing, which can constrict this tubing to reduce inflow. CBI rate is also dependent on the height of the irrigation bags above the bladder because irrigant flow is gravity dependent. Generally, CBI rate should ensure the urine is a dilute pink

and can be slowed or stopped if the urine is clear. Red urine with clots should prompt the CBI rate to be increased.

The stable patient with an appropriately draining bladder is now assessed for the etiology of hematuria and appropriate disposition.

Generally, the stable gross hematuria patient without any clots should be fairly safe for discharge and timely outpatient workup. At least a urine culture should be obtained and a mechanism should be in place to follow-up culture results to ensure prompt antibiotic treatment in the event of a positive result. Those with urinary symptoms or features on urinalysis concerning for infection may be given empiric antibiotics for urinary organisms, which include gram-negative bacteria +/- enterococcus, and staphylococcus. Even with an empiric diagnosis of a urinary tract infection (UTI), it is still imperative that the patient be referred to a urologist for workup as this diagnosis is often revised when culture results return and an underlying malignancy may also be present. The risks and benefits of interrupting anticoagulation for co-existent conditions should be discussed and the prescribing physician notified. Patients on anticoagulation still require a full hematuria workup [1]. Patients can be counseled that their outpatient workup will generally include a detailed history and physical examination, repeat urinary testing, and additional workup such as a flexible cystoscopy under local anesthetic and upper tract imaging [1, 10]. Specific reasons to return to the ED include symptoms of worsening anemia, difficulties urinating due to urinary clots, and symptoms of worsening infection.

Even if a patient has presented in urinary retention from clots, they can sometimes still be discharged for an outpatient workup. Generally, this is in the setting of a limited resolved amount of bleeding that has resulted in clot-related obstruction but is now clear after catheter placement and limited manual and/or continuous bladder irrigation. Patients are generally discharged with an indwelling urethral catheter and short-term urological follow-up.

Patients with clot retention and significant acute renal insufficiency should be observed for possible postobstructive diuresis (POD). POD can be life threatening and is discussed further in the section on urinary retention. It is very hard to estimate urine output in the patient on CBI, and POD monitoring will generally require additional clinical monitoring and serum electrolytes. Urologic consultation in this setting is advised.

The unstable patient or patient with persistent bleeding will require further inpatient assessment and management. After stability and urinary tract drainage are ensured, the initial steps of management include correcting any underlying coagulopathy and treating coexisting infection empirically. The source of bleeding in these patients should be determined. Sometimes the source of bleeding is readily apparent given the history or physical examination. Urology consultation should then be obtained, as most sources of bleeding cannot be controlled in the ED. A discussion of selected oncologic etiologies is detailed below. If the source of bleeding is not readily apparent, differential diagnosis can broadly be categorized as upper or lower urinary tract. The best initial imaging test is a triphasic CT scan with noncontrast, early-contrast, and delayed-contrast images [10]. This test is sometimes called a CT urogram or CT intravenous pyeloureterogram. This will allow for an assessment of a variety of sources of bleeding in the upper and lower urinary tracts. Generally, most upper urinary tract causes can be excluded by imaging, while lower urinary tract etiologies are sometimes not readily apparent and findings within the bladder can be obscured by bladder clot. When a timely CT urogram is contraindicated (e.g., severe contrast allergy, renal failure, lack of availability, or an unstable patient), urology consultation is most appropriate to decide on the best initial imaging modality. Depending on the situation, this may include a renal ultrasound, MR urogram, noncontrast CT, or deferring upper tract imaging if a source is known [10].

Management of the specific source of bleeding will depend on its etiology. A discussion of selected oncologic etiologies of hematuria is detailed below.

Renal Mass

Although most renal masses are detected incidentally, a proportion will still present with hematuria, flank pain, nonspecific abdominal or back symptoms, or a palpable mass [11]. Generally, larger and more centrally located masses are more likely to cause hematuria. An antecedent history of trauma may have precipitated the hematuria but it usually occurs spontaneously. Hematuria associated with renal masses can be associated with ipsilateral flank pain, particularly if there is an ipsilateral perirenal hematoma or clot-related urinary obstruction. Paraneoplastic symptoms may also be present [12]. Most large renal cortical masses will be renal cell carcinoma [13], and masses involving the collecting system are likely to be urothelial carcinoma [14]. One should be wary of assuming that a small peripherally located renal mass or cyst is the cause for hematuria as these are common incidental findings. In some settings, an associated retroperitoneal hematoma may be of higher concern than the hematuria itself. A spontaneous retroperitoneal hematoma requires follow-up imaging, as sometimes the renal mass where the bleed originated is not initially visible amidst hematoma. In rare cases, spontaneous intra-tumor hemorrhage may cause significant pain or gross hematuria. Renal artery embolization can be considered after urologic consultation. Large renal masses should also prompt careful consideration of the presence of intraluminal renal vein or inferior vena cava tumor involvement, which can be present in up to 30% of patients undergoing radical nephrectomy [15] (Fig. 31.2). Vascular involvement may not be readily apparent if a noncontrast CT was obtained or if imaging identifying the mass is old. A further discussion of tumor thrombus is found later in this chapter.

Renal mass-related hematuria often requires intervention. In surgical candidates, this generally involves partial or radical nephrectomy or nephroureterectomy. A stable patient with limited hematuria may be discharged and have nephrectomy following a comprehensive outpatient workup. Significant ongoing hematuria may require a more timely nephrectomy. Nonsurgical candidates are often managed by angioembolization. In metastatic renal cell carcinoma or upper tract urothelial carcinoma, systemic therapy [16] and/ or radiation [17] can also be of assistance in managing hematuria in nonsurgical candidates.

Ureteral Mass

Ureteral masses are often due to urothelial carcinoma [14]. A benign etiology such as a clot from hematuria originating in the ipsilateral kidney, papillary necrosis, or a radiolucent stone may mimic a ureteral mass. Hematuria associated with a ureteral mass is generally managed surgically. The degree of hematuria from a ureteral mass is usually limited and unlikely to require emergent intervention. Concomitant ipsilateral ureteral obstruction will often result in hydronephrosis. Obstructive uropathy and pain, renal dysfunction, or infection should be ruled out prior to outpatient assessment. Management of ureteral obstruction is discussed in more detail below.

Bladder Mass

Bladder masses diagnosed on imaging or prior cystoscopy are most commonly urothelial carcinoma [18]. Clot, mucosal folds, or debris in the bladder however can often mimic a bladder mass and accurate determination can be challenging radiologically. The patient with limited hematuria and a bladder mass is safe for outpatient urologic assessment. A urine culture should be sent to facilitate outpatient workup. A brief period of time until their assessment to await resolution of hematuria may actually be beneficial, as this will improve visualization during office flexible cystoscopy. For those with clot urinary retention, initial management is similar to undifferentiated hematuria as described above with manual followed by continuous bladder irrigation. The stable patient with timely resolution of their hematuria and a limited clot burden can be discharged for an outpatient assessment with an indwelling urethral catheter or following a trial of void. Ongoing hematuria or those with a significant clot burden



Fig. 31.2 Renal cell carcinoma with inferior vena cava tumor thrombus. A 71-year-old female presented to the ED with worsening back and abdominal pain, hematuria, and shortness of breath. She had a mild acute kidney injury along with normal liver function testing. She had an urgent CT pulmonary angiogram, which revealed no evidence of pulmonary embolus. (a) Abdominal CT scan revealed a large right renal mass (not pictured) along with a tumor thrombus in the inferior vena cava veins (*). (b) Echocardiography revealed that the tumor thrombus (*) extended to the level of the right atrium. (c) Echocardiography also revealed that the hepatic veins were free of tumor. Staging investiga-

tions were otherwise negative, and her severe symptoms prevented discharge. Because it was evident on CT that the inferior vena cava thrombus was comprised of tumor rather than bland thrombus, therapeutic anticoagulation was not initiated. She had a right radical nephrectomy and tumor thrombectomy with cardiopulmonary bypass shortly after admission. (d) The pathologic specimen from radical nephrectomy and tumor thrombectomy is pictured with a large right renal mass, adrenal gland, and tumor thrombus (*). Pathology revealed pT3c clear cell renal cell carcinoma with 0 of 33 lymph nodes involved. Expected long-term cancer-specific survival is above 50%

likely warrant inpatient urologic assessment. The management of a patient with suspected bladder mass on continuous bladder irrigation is usually to await resolution for a period of time followed by inpatient cystoscopy if resolution does not occur. Some patients with a bladder mass and very significant hematuria may be taken urgently to the operating room for cystoscopy under general anesthetic to evacuate the clots, control bleeding, and resect the bladder mass for diagnosis and staging. Patients with an advanced bladder mass and recalcitrant hematuria may require numerous interventions, including focused bladder radiotherapy.

Urethral Pathology

A variety of urethral pathology may present to the ED including tumors, strictures, trauma, and infections. Urethral pathology will present with bleeding, urinary symptoms, and a mass. Limited urethral bleeding will present as hematuria at the beginning of urinary stream, which resolves as the stream continues. Significant urethral bleeding will manifest as ongoing hemorrhage that continues outside of micturition if the source is distal to the urinary sphincter. Lower urinary

tract obstruction may result in urinary retention or symptoms such as hesitancy, intermittency, slow stream, frequency, urgency, incomplete emptying, dysuria, post-void dribbling, and overflow incontinence.

For patients in urinary retention, the appropriate catheter placement technique will depend on the pathology. Urology should be involved prior to catheter attempts in patients with suspected external or iatrogenic urethral trauma to avoid worsening the degree of urethral trauma [7]. Often, these patients present with blood at the meatus or gross hematuria that clears during the void. In the patient with a suspected urethral stricture or bladder neck contracture after radical prostatectomy, a smaller (12–14 Fr) catheter should generally be used. Upfront cystoscopy for catheter placement may be required if initial gentle attempts fail and an inability to catheterize the urethra may require suprapubic tube placement.

Women in urinary retention may be especially difficult to catheterize when a urethral or vulvar lesion obscures the native urethral meatus or when prior perineal resection or radiation results in meatal retraction and/or stenosis. Lithotomy position, a speculum blade to retract the posterior vaginal wall posteriorly, additional lighting, and an assistant to part the labia will help with difficult female catheterization. Some women may require a catheter be passed by palpation alone without visualization along the anterior vaginal wall when the urethral meatus is not visible. Additionally, ongoing bleeding and/or severe symptoms may require semiurgent removal of a urethral lesion in women (e.g., strangulated urethral prolapse or caruncle).

Successful catheter placement in a patient with urethral pathology will often lead to clear urinary drainage because the eyelet of an appropriately placed catheter sits in the bladder, proximal to the site of a potentially bleeding urethral lesion. Ongoing urethral bleeding following catheter placement in a patient with urethral pathology may manifest as bleeding around the catheter. This bleeding may be controlled by gentle traction on the catheter along with compression of the distal urethra around the catheter for a period of time to allow for tamponade. This tamponade occurs when a closed system is created by the balloon of the Foley catheter on gentle traction at the bladder neck and the compressed urethra distally. In men requiring ongoing compression, gentle wrapping of the penis with a strip of gauze or Coban dressing can be helpful. A tourniquet effect resulting in penile ischemia and necrosis can result if the penile compression is overly restrictive, and so any dressing placed must be slightly loose and should be removed once the bleeding is controlled. Vigilant penile status monitoring is required if a penile tourniquet is used to avoid ischemic complications.

Once the acute presentation of a urethral lesion is managed, further evaluation will generally be handled as an outpatient. This workup includes a cystoscopy with a biopsy. Imaging for urethral pathology is best approached with MRI due to its enhanced soft-tissue resolution [19].

Urinary Retention

Urinary retention refers to an inability to empty the bladder appropriately. This is common in patients with cancer and may be an acute or chronic finding. The etiology of urinary retention may be an obstruction of the bladder outlet (i.e., the prostate or urethra) or from insufficient contraction of the bladder itself. The specific etiology for urinary retention is often multifactorial and both the bladder and outlet may play a role. Elucidating the factors contributing to urinary retention generally fall into the realm of outpatient urology.

Chronic urinary retention refers to a longstanding inability to empty the bladder. While this may manifest as recently worsening urinary symptoms, patients do not endorse an inability to void or acute pain. Chronic urinary retention is diagnosed by an elevated post-void residual. Post-void residual may be measured by portable bedside ultrasonography [20], but a variety of false positives exist for these machines including obesity, ascites, peritoneal dialysis, malignancy, and anatomic abnormalities such as bladder diverticuli and pregnancy. Post-void catheterization with measurement of the catheterized output is an alternative way to accurately measure post-void residual [20]. There is also no clear agreement on what is an abnormal post-void residual, but chronic urinary retention is often considered for PVRs above 300– 400 ml [21]. Chronic urinary retention is generally assessed on a nonurgent outpatient basis unless an acute complication exists. Decompensation of chronic urinary retention may result in acute urinary retention, UTI, and renal failure. Generally, these issues are managed initially with urethral catheterization. Subsequent management is coordinated by urology and involves discharge and close follow-up in many cases.

Acute urinary retention refers to an acute inability to empty the bladder. Typically, patients have significant difficulty or total inability to void, pelvic pain, and overflow incontinence, which does not relieve their symptoms. They may have baseline urinary symptoms that have worsened recently including a weak stream, straining to void, incomplete emptying, hesitancy, intermittency, urgency, and frequent small volume voids. Those with a preceding or complicating infection may also have dysuria, fevers, flank pain, or testicular symptoms. Hematuria is a frequent presenting symptom for those with clot urinary retention. Signs include abnormal vital signs, a visually uncomfortable and restless patient, abdominal distension, and a palpable bladder. Other abnormalities on the genitourinary exam may be associated with the cause of the urinary retention. The emergency physician should have a low threshold to suspect acute urinary retention in any patient presenting with pelvic discomfort or urinary complaints.

A clinical diagnosis of acute urinary retention is usually confirmed by an elevated post-void residual measured by bedside ultrasonography [20]. If this is not immediately available, the patient should be catheterized empirically if acute urinary retention is suspected. A high volume of output upon catheter placement along with resolution of symptoms will confirm this diagnosis. Following catheter placement, the patient should be assessed for conditions associated with urinary retention including infection, hematuria, and renal failure. If isolated urinary retention is present and symptoms have resolved following catheter placement, the patient may be discharged with an indwelling urethral catheter for an outpatient workup and trial of voiding.

The amount of urine output following relief of urinary tract drainage is important. It is normal to have significant output upon initial relief of urinary tract obstruction (i.e., several hundred milliliters up to several liters in a short time period may be normal). If the urine output is less than expected, consideration should be given to an alternate diagnosis or to a mispositioned or obstructed catheter. Alternatively, a post-obstructive diuresis (POD) can result following urinary tract drainage [22]. A patient is considered to have POD when they have a persistently high urine output (usually >200 mL/h) following relief of obstruction. This generally occurs following relief of obstruction in a patient with urinary retention that has resulted in renal failure. A similar phenomenon occurs following relief of bilateral ureteral obstruction or unilateral ureteral obstruction in a patient with a solitary kidney. A certain degree of POD is physiologic as longstanding obstruction and/or renal failure may have resulted in fluid retention requiring excretion following relief of obstruction. Pathologic POD may continue beyond this physiologic period resulting in fluid status and electrolyte abnormalities that may be life threatening.

Post-obstructive diuresis should be suspected in any patient presenting with retention and significant renal insufficiency. These patients should have urine output, volume status, vital signs, and electrolytes monitored following relief of their urinary tract obstruction. Patients without mental status changes should be provided free access to water, as thirst will allow them to self-regulate their fluid status. Aggressive fluid replacement should be avoided as this can prolong diuresis. The patient with limited polyuria, no electrolyte abnormalities, normal vital signs, normal volume status, and adequate mentation to drink when they are thirsty should be safe for discharge after a period of observation (e.g., 6 h). Patients not meeting these criteria will generally require admission for monitoring. In patients with a severe diuresis or who lack the ability to self-regulate their fluid status, fluid replacement is generally administered at a rate of half the urine output with 0.45% normal saline. Patients with severe renal failure, electrolyte abnormalities, or fluid status imbalance will benefit from nephrology consultation and modification of their fluid replacement based on urinary and/or serum electrolytes.

Hydronephrosis

Hydronephrosis refers to dilation of the collecting system of the kidney including the renal pelvis and calyces. This may be associated with ureteral dilation or hydroureter. The combination of renal and ureteral dilation is termed hydroureteronephrosis. The etiology of hydronephrosis may be obstructive or nonobstructive.

The first consideration in any patient with hydronephrosis is whether it may relate to urinary retention due to a lower urinary tract pathology. This will present with a distended bladder and bilateral hydroureteronephrosis. A urethral catheter will relieve lower urinary tract obstruction and should be placed expediently for patients with acute urinary retention, concern for infection, or renal failure. Concomitant bilateral ureteral obstruction may be present, and this should be considered in the patient with urinary retention and ongoing hydronephrosis, oliguria, signs of infection, or renal failure.

Although nonobstructive hydronephrosis may be relevant in certain populations, most hydronephrosis in the cancer population is obstructive. This is because ureteral obstruction by abdominopelvic malignancy is a fairly common complication of advanced malignancy [23]. The specific etiology of upper urinary tract obstruction may be intraluminal or extraluminal. Common intraluminal etiologies include ureteral tumors, stones, strictures, or an encrusted indwelling ureteral stent. External compression may result from any number of different intraabdominal pathologies. Malignant etiologies such as retroperitoneal or pelvic lymphadenopathy or a pelvic mass are fairly common [23]. Otherwise, ureteral obstruction may be iatrogenic, post-traumatic, inflammatory, infectious, congenital, or idiopathic. Although the differential diagnosis of ureteral obstruction is lengthy, the main consideration for the emergency physician can be simplified into whether urgent decompression is required or not.

It is also important to understand that urinary tract obstruction may also be present without significant hydronephrosis [24]. This is particularly true in the setting of acute obstruction when hydronephrosis may not have had a chance to develop. Accordingly, the patient with a clinical history suggestive of obstruction, even with limited or absent hydronephrosis, should be evaluated by a urologist (e.g., the patient with anuria despite urethral catheterization and radiographically evident pathology that may cause ureteral obstruction on CT). Moreover, changes in the extent of hydronephrosis between interval studies is difficult to quantify and interpret and can vary based on numerous factors other than the degree of obstruction [25].

The indications for urgent upper urinary tract decompression are a concern for obstructed UTI, renal failure, and uncontrolled flank pain. The emergency physician should have a low threshold to suspect infection in a patient with hydronephrosis and signs of sepsis or a positive urinalysis. The urinalysis may be falsely negative as a complete obstruction may disconnect the obstructed urinary tract from voided urine. Given the gravity of missing an obstructed UTI, the threshold should be low to consult urology in the setting of hydronephrosis with a concern for infection. Significant recent worsening of renal insufficiency, particularly with associated oliguria, should also be considered an indication for urgent decompression. It is sometimes challenging to elucidate whether a minor degree of acute kidney injury relates to limited hydronephrosis in the acute setting. When in doubt, a urology consultation is reasonable, but consultants may elect to observe this hydronephrosis in the stable patient and trend their renal function. Ongoing symptoms despite analgesia are also an indication for upper urinary tract decompression.

Upper urinary tract drainage is accomplished by ureteral stenting or nephrostomy tube placement. In the infected patient, randomized data have not supported any difference in meaningful outcome such as time to fever resolution [26]. Ultimately, the decision on which to perform is very nuanced and dependent on patient factors. In the patient requiring urgent intervention, the best measure will depend on what is available first. It is known that there is significant regional variation in stent versus nephrostomy tube placement likely for this reason [27]. Contraindications to each decompression method are first assessed. Nephrostomy tubes will generally be contraindicated in those with coagulopathy and may be more difficult in the patient who cannot be placed prone, has morbid obesity, or lacks hydronephrosis. Ureteral stents will not be feasible in the patient who does not have easy endoscopic access to the ureter-in the oncologic setting this is due to trigonal invasion of a large pelvic mass or due to prior lower urinary tract surgery. Additionally, suspicion of longstanding complete obstruction will be very difficult to bypass with a ureteral stent and consideration should initially be made for nephrostomy tube placement. While a ureteral stent can be placed under local anesthetic, it is often painful and usually performed with sedation or a general anesthetic. Anesthetic requirements are likely less for nephrostomy tube placement and this procedure can be conducted at the bedside in an emergency if necessary.

If no specific factors indicate one procedure over the other, the choice for nephrostomy tube versus ureteral stent is generally undertaken by shared decision-making. Ultimately, patient and practitioner factors heavily influence the choice of urinary diversion modality [28].

Following decompression of the upper urinary tract by either stent or nephrostomy tube, it is important to consider the possibility of post-obstructive diuresis. This will generally occur in the setting of significant renal insufficiency associated with the relieved obstruction. This is discussed further in the setting on urinary retention.

One particular etiology of obstruction that warrants discussion in the oncologic patient is stent failure. Abdominal malignancies often cause ureteral compression requiring ureteral stent placement. Due to progressive growth of these malignancies or encrustration of an indwelling stent over time, recurrent obstruction can develop despite an indwelling stent. The diagnosis of stent failure can be hard to assess in the ED as degree of hydronephrosis is not necessarily indicative of impaired drainage. Generally, stent obstruction should be suspected if the patient has recurrent clinical sequelae of obstruction. One should look for signs of obstructed infection, renal dysfunction, worsening hydronephrosis, and acute or subacute worsening of flank pain. This diagnosis of stent obstruction can sometimes be difficult to make as indwelling stents or the underlying obstructive pathology may cause urinary symptoms or flank discomfort.

Theoretically, renography could assist in the diagnosis, but it is often not necessary. Management is generally either with a stent exchange or conversion to nephrostomy tube with subsequent stent removal. The latter approach is warranted particularly if re-obstruction occurs quickly, a stent is poorly tolerated, or stent exchange is complicated.

Although nonobstructive hydronephrosis is possible, this is a difficult initial diagnosis to make in the ED. The diagnosis generally requires diuretic nuclear renography to rule out obstruction. Nonetheless, nonobstructive hydronephrosis may be evident from the history, particularly if diuretic renography has previously demonstrated a lack of obstruction. Possible nonobstructive etiologies include vesicoureteral reflux, pregnancy, excessive hydration, congenital anomalies, and prior corrected obstruction. A parapelvic renal cyst may also often masquerade as hydronephrosis [24]. Patients with history of radical cystectomy with urinary diversion can have physiologic hydronephrosis without obstruction and a loopogram or pouchogram will be diagnostic. Because of the relative infrequency of nonobstructive hydronephrosis and the difficulty in making this diagnosis, it would be reasonable to consider most new hydronephrosis in the ED as obstructive.

Testicular Cancer

Testicular cancer is the most common solid organ malignancy in men aged 15-44 years [29]. As men in this age group infrequently see a healthcare practitioner in the outpatient setting, they often have delayed diagnosis [30-32] and may present symptomatically to the ED. Presenting symptoms include a testicular mass, enlargement, and discomfort (Fig. 31.3). They may also have symptoms of systemic disease including abdominal or back pain, respiratory symptoms or hemoptysis, breast enlargement or tenderness, or neurologic symptoms [33]. The physical exam will usually reveal a palpable testicular mass with ipsilateral hemiscrotal enlargement. A reactive hydrocele, discomfort, contralateral or multifocal masses are also sometimes present. The genitourinary examination is often otherwise unremarkable. An abdominal mass may be present in the setting of bulky retroperitoneal adenopathy. Gynecomastia may also be present in the setting of elevated tumor markers or a testicular stromal tumor. The respiratory exam may be abnormal due to thoracic metastases. A Virchow's node (supraclavicular) may be palpable along with additional neck adenopathy. Patients will rarely also have symptoms from brain metastases, bone metastases, venous obstruction, or thromboembolism [34]. The differential diagnosis for testicular masses includes epididymoorchitis with possible abscess formation, testicular torsion, hernias, paratesticular lesions, hydroceles, varicocele, inflammatory conditions, and trauma. Various other



Fig. 31.3 Testicular tumors. These patients presented initially with scrotal complaints, were imaged with a scrotal ultrasound, and ultimately underwent radical orchiectomy. (a) This poorly circumscribed hypoechoic mass was found to be lymphoma. (b) This well-circumscribed, small, and difficult-to-palpate lesion was found to be a Leydig cell tumor. (c) Some testicular tumors have significant hypoechoic areas consistent with cystic components—this finding is suggestive of teratoma, which comprised the majority of this patient's nonsemonimatous germ cell tumor. (d) Gynecomastia is sometimes associated with testicular tumors. This can occur due to hormonal secretion from testicular stromal tumors (in this patient estradiol was being secreted by the Leydig cell tumor depicted in b). Gynecomastia can also occur due to elevated beta-HCG secreted by germ cell tumors. (e) This patient had increasing abdominal pain and leg swelling during

malignancies or conditions can cause the systemic findings detailed above. Although the differential diagnosis is broad, testicular cancer should always be considered for any scrotal complaint in the young man.

Initial workup for a testicular complaint should include scrotal ultrasonography to confirm the presence of a testicular mass with a urinalysis to rule out infection [35] (see Fig. 31.3). The mass will generally appear as a circumscribed hypoechoic hypervascular lesion on sonography that is within the testicle itself. The presence of blood flow on Doppler sonography will assist in ruling out testicular torsion, although cases of torsion may be present with vascularity. Although these findings are characteristic, testicular cancer can sometimes be present without visible flow on Doppler or a visible mass [35]. A negative urinalysis will generally rule out an infectious etiology, but additional

chemotherapy for nonseminomatous germ cell tumor. He had multiple presentations to the ED for pain control. He was eventually diagnosed with growing teratoma syndrome as his large retroperitoneal mass was expanding during chemotherapy. This mass also compressed and laterally displaced the inferior vena cava. The diagnosis of growing teratoma syndrome should be suspected in the patient presenting with worsening symptoms during chemotherapy for nonseminomatous germ cell tumor. Imaging will reveal progressive growth of metastatic disease from pre-chemotherapy scans. In this patient, chemotherapy was interrupted and the patient was taken for surgical resection. Pathologic findings from surgery revealed pure teratoma. (\mathbf{f} , \mathbf{g}) Germ cell tumors can be hypo-, iso-, or hyperechoic, and can sometimes be difficult to distinguish from surrounding normal parenchyma

sexually transmitted infection testing may be indicated in certain scenarios. Complete blood counts, a complete metabolic panel, and testicular tumor markers (alpha-fetoprotein [AFP], beta-human choriogonadotropin [beta-HCG], and lactate dehydrogenase [LDH]) should also be sent. Generally a high AFP (>20 ng/mL) or beta-HCG along with a characteristic testicular mass will strongly suggest the diagnosis of testicular cancer. Those with extra-testicular signs or symptoms warrant emergent imaging with a CT scan of the chest, abdomen, and pelvis. Extremity complaints or concern for pulmonary embolism should also prompt extremity duplex and/or CT pulmonary angiography. Neurologic complaints or a very elevated beta-HCG (>5000 IU/mL [36]) warrant brain imaging or spine imaging with MRI. Patients with an equivocal diagnosis or without any extra-testicular complaints can have staging imaging deferred to the outpatient setting where it may be omitted if they are diagnosed with a benign lesion. Alternatively, a chest X-ray rather than CT may be sufficient in staging patients with suspected stage I seminoma [35].

If the patient has systemic symptoms or significant metastatic disease burden, a conversation with a urologic oncologist or genitourinary medical oncologist prior to discharge is warranted. In this uncommon scenario, urgent inpatient treatment is sometimes required and may be undertaken based on a presumptive diagnosis from the clinical scenario and serum tumor markers even without a formal tissue diagnosis [36]. It is more common that a new testicular cancer patient will be assessed as an outpatient with close interval follow-up with a urologist. In general, biopsy of retroperitoneal lymph nodes is not pursued and a prompt radical orchiectomy is sufficient for diagnosis.

Retroperitoneal Mass with Venous Involvement

Renal cell carcinoma is associated with venous involvement in 30% of patients undergoing radical nephrectomy [15]. Other retroperitoneal tumors, including adrenal, upper tract urothelial, sarcoma, and testicular cancers can also involve the venous system. The level of venous involvement can range from thrombus within the small intrarenal vessels alone to complete involvement of the renal veins, inferior vena cava, and right atrium. Patients are often symptomatic, and the degree and pattern of morbidity will depend on their level of venous involvement (see Fig. 31.2). One should have a high degree of suspicion for venous involvement in the patient with a known or suspected renal mass presenting with worsening symptoms. Symptoms can include hematuria, lower extremity edema, ascites, abdominal flank or back pain [11], and a variety of other symptoms associated with generalized malignancy, renal failure, hepatic failure, or cardiac failure. Physical exam and lab findings are variable and will also associate with degree of impact on various organs. The diagnosis of venous involvement is generally made by CT scan with IV contrast. Pulmonary embolism should be ruled out with a CT pulmonary angiogram in any patient presenting with respiratory symptoms. Budd-Chiari syndrome must be considered in the patient presenting with a thrombus close to the hepatic veins [37]. An echocardiogram will be helpful in assessing the patient with intra-atrial thrombus and evidence of heart failure. Ultimately, urologic oncologic consultation should be obtained to develop a plan for the patient prior to considering discharge. Anticoagulation is not necessarily indicated if venous involvement relates to tumor thrombus rather than bland thrombus and should be started as a shared decision with urologic oncologic [38]. Particularly

in the case of intracardiac thrombus, use of heparin should be discussed with urologic oncologic prior to initiation because the development of heparin-induced thrombocytopenia may prevent safe cardiopulmonary bypass [39].

The first consideration for a patient with a venous tumor thrombus is whether they are a surgical candidate for tumor thrombectomy and the timeline in which this operation needs to occur (see Fig. 31.2). Depending on the primary malignancy and degree of symptoms, these patients may be admitted for symptom control, workup, and timely inpatient intervention. This is a complex problem, and some patients may not be candidates for surgery due to patient or malignancy-related factors. In these situations, systemic therapy can sometimes be effective [40].

Penile Cancer

Penile cancer is rare with 2200 cases a year in the United States [41]. Nonetheless, the delayed presentation of this cancer is well documented [42, 43]. It is essential that this diagnosis be considered in any patient presenting with a penile lesion (Fig. 31.4). Clinical presentations can range from a subtle asymptomatic ervthematous plaque that may be associated with carcinoma in situ to a large fungating penile mass causing urethral obstruction. Patients with a "rash" of the genitals are frequently treated with antifungals and steroids for a prolonged period prior to delayed diagnosis of penile cancer [42]. Additionally, one should also have a low suspicion for the patient with a condyloma to have penile cancer as the enlarging vertucous lesion is also a fairly common presentation and penile cancer is associated with HPV in 40% [44]. When in doubt, urology or dermatology consultation should be obtained. A patient with a penile mass involving the urethra may sometimes present with urinary retention or significant urinary symptoms. Urethral catheterization can be attempted, but the meatus may not be visible amidst a large mass on the glans penis. In some situations, this will require suprapubic catheterization by urology or interventional radiology. Patients with penile cancer may also present with superinfection of a necrotic penile mass, including Fournier's gangrene, albeit an infrequent situation.

Another acute presentation associated with penile cancer is inguinal adenopathy. The first site of metastasis from penile cancer is to the inguinal lymph nodes [45]. These are generally palpable when they are involved in the nonobese patient (see Fig. 31.4). There are also some situations of bulky inguinal adenopathy without a diagnosed primary, which may ultimately lead to a diagnosis of anal, urethral, vulvar, or lower extremity cancers that drain into these nodal basins. The patient with asymptomatic inguinal adenopathy is likely safe for an outpatient assessment. However, in some situa-



Fig. 31.4 Penile cancer. (**a**, **b**) Inguinal adenopathy associated with penile cancer can lead to various complications. In (**a**) a necrotic lymph node has eroded through the skin and resulted in a nonhealing wound and ongoing purulent discharge. Bulky inguinal adenopathy can also erode into the femoral vessels. In (**b**) there is no separation between the bulky right inguinal lymph node and either the femoral artery or vein. This can eventually result in life-threatening hemorrhage. (**c**) Sometimes a penile mass is not immediately evident as in this patient with penile squamous cell carcinoma who did not look at the ventral aspect of his penis in several years. (**d**) Penile cancers can also be limited in size relative to associated adenopathy. Here a small squamous cell carcinoma is seen on the penile shaft. Bulky inguinal adenopathy is evident on exam.

tions inguinal adenopathy may erode into the femoral vessels, which can cause massive hemorrhage [46, 47]. Urgent urologic oncologic and vascular surgery consultation should be obtained in this case. Additionally, patients with inguinal adenopathy present with severe leg, perineal, or penoscrotal lymphedema, with pain and skin changes. Superimposed infection and deep vein thrombosis are important to rule out. Primary cancer treatment along with symptom control, palliative care consultation, and lymphedema clinic referral can all be useful.

Prostate Cancer

Prostate cancer is the most common cancer in American men with 174,650 cases in 2019 [48]. Although many prostate cancers are localized, metastatic disease is still a large cause of morbidity, and it is the second most common cause of This patient's pelvic CT scan is depicted in panels (**a**), (**b**), and (**e**). Cancers can rarely metastasize to the penis as in this patient with renal cell carcinoma involving the glans penis. (**f**, **g**) Penile cancers may initially be confused for a benign process. The patient in (**f**) had a circumcision for what was thought to be chronic balanitis. The pathology from this circumcision revealed squamous cell carcinoma. He then needed a partial penectomy and inguinal lymph node dissection. This patient in (**g**) was treated with steroids for months due to a penoscrotal rash. Progression of the lesion prompted a biopsy revealing extramammary Paget's disease. He required excision of all the superficial tissues of his penis and scrotum

cancer death in America with 31,620 deaths in 2019 [48]. Generally, prostate cancer can present to the ED in three ways.

- (i) Urinary retention and/or hematuria. The ED management of these conditions is addressed above. Patients should be referred to a urologist for further management.
- (ii) Upper urinary tract obstruction is also common in advanced prostate cancer. This is also discussed above. Patients should be referred to a urologist for further management.
- (iii) Spinal cord compression. Prostate cancer is the second most common etiology of malignant spinal cord compression [49]. Those with bony metastatic disease can experience bony pain, pathologic fracture, and spinal cord compression. Patients with cord compression require timely management to minimize permanent

neurologic sequelae [50]. The crucial difference in managing prostate cancer cord compression compared to other cancer sites is the need to institute androgen deprivation therapy. It has been known since the 1940s that prostate cancer is sensitive to the lowering of serum testosterone, which induces atrophy in prostate cancer cells [51]. In the setting of cord compression, a tissue diagnosis of prostate cancer is not required to institute androgen deprivation therapy. Nonetheless some clear evidence of prostate cancer should be present such as a significantly elevated PSA, osteoblastic metastases, and an abnormal rectal examination. Rapid reduction of testosterone is essential in the setting of cord compression. The conventional method of doing this in the outpatient setting is to start the patient on an androgen receptor blocker (e.g., bicalutamide) along with delayed administration of GnRH agonist. This is not appropriate in suspected cord compression because these methods take a long time to work and may induce a testosterone flare [52]. Instead patients must be started on either a GnRH antagonist [52], ketoconazole, or have an urgent bilateral orchiectomy. Other than androgen deprivation therapy, management of prostate cancer cord compression has the same basic principles as other cancer types. including complete spine imaging, high-dose steroids, and early involvement of spine surgery and radiation oncologic consultants. The prostate cancer patient can sometimes present with spinal cord compression after having received multiple prior lines of systemic therapy with progression of disease. Testosterone is generally already castrate in this situation and androgen manipulation will have little effect.

Postoperative Issues

Prostatectomy

Catheter or Urinary Issues Following prostatectomy, patients will be discharged with a urethral catheter as the vesicourethral anastomosis heals. Patients may present with decreased urine output, catheter discomfort, catheter bypassing, hematuria, or debris in the catheter. Usually the only intervention that should be undertaken in the ED is to consider gently irrigating or flushing the catheter. The catheter should generally not be removed, repositioned, or replaced by anyone other than a urologist because of the recent anastomosis and the potential that this has already disrupted or may potentially be disrupted by catheter manipulation. At times, patients will present with urinary retention or new urinary symptoms following prostatectomy catheter removal. Urinary retention and a UTI should be ruled out. Generally,

patients should not have a urethral catheter inserted unless the issue is discussed with a urologist due to a concern for traumatizing a recent vesicourethral anastomosis. Catheter bypassing and urinary incontinence after prostatectomy catheter removal are normal, and patients should be reassured, provided it is clear that their bladder is empty and they are not retaining urine.

Pelvic Collections Pelvic collections are commonly found in the patient presenting post-prostatectomy who might receive a CT scan for a variety of indications. The differential diagnosis includes lymphocele, abscess, urinoma, seroma, hematoma, or bowel injury. Radiologic characterization is often not possible, and these collections may frequently be described as abscesses because they contain gas, although this appearance can relate to hemostatic products used at the time of surgery (e.g., oxidized cellulose) [53]. Patients with signs or symptoms of an infection require prompt antibiotic treatment, resuscitation, and percutaneous drainage. Those with bleeding, urinoma, or findings concerning for bowel injury require additional procedures. Generally urologic consultation should be obtained following the initial management of a patient with a pelvic collection post-prostatectomy.

Radical Cystectomy

The readmission rate after radical cystectomy (RC) for bladder cancer is 20–30%; thus, these patients are frequently seen in the ED after undergoing surgery [54].

Fever

Fever is one of the most common reasons for readmission after RC as infectious concerns comprise 25% of postoperative complications [54]. The patient with a fever after RC should be worked up broadly. Specific etiologies include pyelonephritis, intraabdominal infections, wound infections, urinary leak, bowel leak or bowel injury, pulmonary embolism or pneumonia, Clostridioides difficile colitis, and line sepsis. Broad workup including chest imaging and abdominal imaging are warranted. Due to bowel being routinely used as part of the urinary diversion, urinalysis and culture may be falsely positive and not indicative of the true source of infection. Patients will often have received multiple antibiotics recently around the time of surgery and prior readmissions; thus, consideration should be given to coverage of MRSA and ESBL in the choice of empiric antibiotic. CT scan of the abdomen and pelvis is essential in the early postoperative period. Timely urologic consultation is advised.

Hydronephrosis

Hydronephrosis is a common finding on a CT scan performed in the patient after RC. The approach to these patients is similar to the general approach detailed above. Strong consideration should be given to prompt nephrostomy placement in the patient presenting with sepsis or renal failure. Retrograde attempts at stent placement are not usually possible.

Urinary Tract Infection

UTIs after RC are relatively common. Symptoms include fever, chills, and flank pain. Presentation can be insidious in the patient presenting with lethargy and mental status changes. Early antibiotic treatment and resuscitation is essential. Urinalysis and culture can both be misleading as this will usually be positive in a patient with a urinary diversion even in the absence of UTI. The urinary findings will have to be correlated with the clinical picture to make a diagnosis, and urology consultation can be helpful in making this determination. Concern for UTI should prompt abdominal imaging, and hydronephrosis generally requires nephrostomy placement. Although hydronephrosis may sometimes be physiologic in the patient with a urinary diversion, the consequences of missing this source of infection in an obstructed UTI is usually an indication for nephrostomy.

Dehydration

Gastrointestinal issues are the most common complication after RC and associated dehydration is common [54]. Patients will present with malaise, fatigue, reduced appetite and oral intake, diarrhea, nausea, vomiting, and reduced urine output. Labs will often reveal an increased creatinine and electrolyte abnormalities. Intraabdominal pathology should be ruled out; the diagnosis of dehydration is one of exclusion. Patients presenting with dehydration will often require admission for rehydration and closer monitoring on discharge to avoid another readmission. Associated gastrointestinal issues should be managed by the surgical team.

Leg Swelling

In addition to the usual differential, the emergency practitioner should be especially suspicious of deep vein thromboses due to a high rate of venous thromboembolism after RC [55]. Additional etiologies include postoperative fluid overload, pelvic fluid collection, and lymphedema related to a pelvic lymph node dissection.

Diarrhea

Diarrhea can be a challenge after RC due to bile acid malabsorption from use of the terminal ileum as part of most forms of urinary diversion. The ileocecal valve may also not be present in the patient with an Indiana pouch. Diarrhea should be differentiated from tenesmus or pelvic discomfort, which can often be secondary to intraabdominal pathology. Nonetheless, patients are at risk of infectious etiologies including *Clostridioides difficile* colitis. Patients presenting with diarrhea should be resuscitated, and infection with dehydration and electrolyte abnormalities ruled out. Antidiarrheals such as loperamide or cholestyramine can be used once infection is ruled out.

Acute Abdomen

Patients presenting with an acute abdomen are managed similarly to other surgical patients. Early surgical consultation is obviously warranted. Etiologies could relate to the gastrointestinal or urinary system and the differential diagnosis is complex [54]. Patients should be resuscitated and if imaging is performed before urologic consultation, a CT urogram is often more useful than a standard CT scan because this will aid in differentiating many complications associated with the urinary diversion. Oral contrast and/or rectal contrast may also be useful depending on the situation.

Stomal Complications

Most patients will have an ileal conduit after RC. Patients may develop various issues with these conduits, including pouch leakage, parastomal hernias, parastomal dermatitis or infections, or issues with the stoma itself (e.g., bleeding, varices, stenosis, prolapse, and necrosis). Stomal bag-related issues and parastomal dermatitis are often best handled by involving a wound ostomy continence nurse to help with troubleshooting and finding the best ostomy supplies. Stomal stenosis will present with a distended conduit on imaging along with oliguria and renal dysfunction; management is generally by stomal catheterization and surgical consultation is advised. Patients with portal hypertension may present with stomal varices, and management is similar to varices elsewhere in the gastrointestinal tract. Blood loss anemia is common in this clinical context. Parastomal hernias are very common on imaging, present in 50% of patients [56]. Although parastomal hernias may be dramatic on the physical examination, these are generally not an acute issue in the ED unless hernia-related complications develop. If there is concern for obstruction or strangulation associated with a parastomal hernia, early surgical consultation is advised.

Continent Diversion Problems

Continent urinary diversion includes both continent cutaneous diversions and orthotopic neobladders. Many types exist and the exact configuration of the urinary diversion is often unclear to the emergency physician. Patients with continent cutaneous diversions will have some form of catheterizable abdominal stoma, while those with an orthotopic neobladder will void per urethra. Various issues with these complicated urinary diversions will cause patients to present to the ED. The emergency provider should seek urologic consultation in devising the management plan and always keep urinary retention or diversion rupture within the differential diagnosis. Generally, procedural interventions should be avoided by the nonspecialist as the anatomy of the urinary diversion is often unfamiliar.

Nephrectomy

Hematuria

Patients presenting after partial nephrectomy with hematuria may develop a pseudoaneurysm or arteriovenous fistula [57]. These patients can present with hemorrhagic shock. In the stable patient, initial resuscitation should be commenced and a pseudoaneurysm/AVM ruled out with CT angiography. In the unstable patient, urologic consultation should be obtained immediately. Management should be coordinated with urology and will often involve selective angioembolization by interventional radiology [58]. Patients should not have hematuria after radical nephrectomy and thus should be assessed as any other patient presenting with hematuria.

Abdominal Collections

Patients after partial or radical nephrectomy may frequently receive a CT scan for abdominal complaints revealing an abdominal fluid collection. The differential diagnosis includes abscess, urinoma, seroma, hematoma, pancreatic leak, bowel injury, or biloma. Generally, initial management includes stabilization with empiric antibiotics if signs of infection are present. The patient should receive nothing by mouth and urologic consultation should be obtained to assess for management of the collection. Many postoperative collections are incidental and asymptomatic but occasionally percutaneous drainage by interventional radiology is required. The rare situation of an acute abdomen suggestive of peritonitis or bleed may warrant an exploratory laparotomy.

Pulmonary Symptoms

The common nonsurgical causes should be ruled out including pulmonary embolism, heart failure, myocardial infarction, and pneumonia. Nephrectomy-specific etiologies can include a pneumothorax, pneumomediastinum, or pneupericardium, which can result from diaphragmatic entry [59], diaphragmatic neuromuscular pathology [60], or difficulty with inspiration due to a nephrectomy bed collection or incisional pain.

Conclusion

Urologic oncologic emergencies present frequently to the ED. Knowledge of urologic anatomy and common presenting issues aids greatly in the triage and initial management of these patients. Most acute presentations will relate to bleeding, obstruction, or infection of the urinary tract. A thorough history and physical exam with pertinent laboratory studies will clarify the presenting complaint, but imaging is frequently essential in diagnosis. Initial management focuses on stabilizing the patient, controlling pain, and ensuring appropriate urinary tract drainage. Early coordination with urologic consultants can be helpful to ensure that appropriate steps are taken in a timely fashion. Once the acute situation has been managed, a full assessment is nuanced and usually best performed in the outpatient setting by a specialist. When an outpatient assessment is appropriate, timely coordination of an appointment is essential to ensure optimal patient care and avoiding repeated ED presentation.

References

- Davis R, Jones JS, Barocas DA, Castle EP, Lang EK, Leveillee RJ, et al. Diagnosis, evaluation and follow-up of asymptomatic microhematuria (AMH) in adults: AUA guideline. J Urol. 2012;188(6 Suppl):2473–81.
- Jung H, Gleason JM, Loo RK, Patel HS, Slezak JM, Jacobsen SJ. Association of hematuria on microscopic urinalysis and risk of urinary tract cancer. J Urol. 2011;185(5):1698–703.
- Boorjian SA, Raman JD, Barocas DA. Evaluation and management of hematuria. In: Partin AW, Peters CA, Kavoussi LR, Dmochowski RR, Wein AJ, editors. Campbell-Walsh-Wein urology. 12th ed. Philadelphia: Elsevier; 2020. p. 16, 247-59.
- 4. Aycock RD, Kass DA. Abnormal urine color. South Med J. 2012;105(1):43–7.
- Linder BJ, Bass EJ, Mostafid H, Boorjian SA. Guideline of guidelines: asymptomatic microscopic haematuria. BJU Int. 2018;121(2):176–83.
- Georgieva MV, Wheeler SB, Erim D, Smith-Bindman R, Loo R, Ng C, et al. Comparison of the harms, advantages, and costs associated with alternative guidelines for the evaluation of hematuria. JAMA Intern Med. 2019;179(10):1352–62.

- 7. Morey AF, Brandes S, Dugi DD 3rd, Armstrong JH, Breyer BN, Broghammer JA, et al. American Urological Assocation. Urotrauma: AUA guideline. J Urol. 2014;192(2):327–35.
- Son HW, Kim DY, Lee OK, Park SE, Cho YW. Severe hypotension following urethral catheterization during general anesthesia in a patient with bladder cancer -a case report. Korean J Anesthesiol. 2012;62(1):91–5.
- 9. Sutera SP, Skalak R. The history of Poiseuille's law. Annu Rev Fluid Mech. 1993;25(1):1–20.
- Expert Panel on Urological Imaging, Wolfman DJ, Marko J, Nikolaidis P, Khatri G, Dogra VS, Ganeshan D, et al. ACR Appropriateness Criteria® Hematuria. J Am Coll Radiol. 2020;17(5S):S138–47.
- Lee CT, Katz J, Fearn PA, Russo P. Mode of presentation of renal cell carcinoma provides prognostic information. Urol Oncol. 2002;7(4):135–40.
- Moreira DM, Gershman B, Lohse CM, Boorjian SA, Cheville JC, Leibovich BC, et al. Paraneoplastic syndromes are associated with adverse prognosis among patients with renal cell carcinoma undergoing nephrectomy. World J Urol. 2016;34(10):1465–72.
- Thompson RH, Kurta JM, Kaag M, Tickoo SK, Kundu S, Katz D, et al. Tumor size is associated with malignant potential in renal cell carcinoma cases. J Urol. 2009;181(5):2033–6.
- Chlapoutakis K, Theocharopoulos N, Yarmenitis S, Damilakis J. Performance of computed tomographic urography in diagnosis of upper urinary tract urothelial carcinoma, in patients presenting with hematuria: systematic review and meta-analysis. Eur J Radiol. 2010;73(2):334–8.
- Blute ML, Leibovich BC, Lohse CM, Cheville JC, Zincke H. The Mayo-Clinic experience with surgical management complications and outcome for patients with renal cell carcinoma and venous tumour thrombus. BJU Int. 2004;94(1):33–41.
- Méjean A, Ravaud A, Thezenas S, Colas S, Beauval JB, Bensalah K, et al. Sunitinib alone or after nephrectomy in metastatic renalcell carcinoma. N Engl J Med. 2018;379(5):417–27.
- Siva S, Ellis RJ, Ponsky L, Teh BS, Mahadevan A, Muacevic A, et al. Consensus statement from the international radiosurgery oncology consortium for kidney for primary renal cell carcinoma. Future Oncol. 2016;12(5):637–45.
- Sadow CA, Silverman SG, O'Leary MP, Signorovitch JE. Bladder cancer detection with CT urography in an academic medical center. Radiology. 2008;249(1):195–202.
- Grivas PD, Davenport M, Montie JE, Kunju LP, Feng F, Weizer AZ. Urethral cancer. Hematol Oncol Clin North Am. 2012;26(6):1291–314.
- Asimakopoulos AD, De Nunzio C, Kocjancic E, Tubaro A, Rosier PF, Finazzi-Agrò E. Measurement of post-void residual urine. Neurourol Urodyn. 2016;35(1):55–7.
- International Continence Society. Urinary retention. Last updated June 2018. https://www.ics.org/committees/standardisation/terminologydiscussions/urinaryretention. Accessed 15 June 2020.
- 22. Halbgewachs C, Domes T. Postobstructive diuresis: pay close attention to urinary retention. Can Fam Physician. 2015;61(2):137–42.
- Donat SM, Russo P. Ureteral decompression in advanced nonurologic malignancies. Ann Surg Oncol. 1996;3(4):393–9.
- Mostbeck GH, Zontsich T, Turetschek K. Ultrasound of the kidney: obstruction and medical diseases. Eur Radiol. 2001;11(10):1878–89.
- Dason S, Braga LHP. Prenatal hydronephrosis. In: Dahm P, Dmochowski R, editors. Evidence-based urology. 2nd ed. Hoboken/ Chichester: John Wiley & Sons; 2018. p. 187–96.
- Pearle MS, Pierce HL, Miller GL, Summa JA, Mutz JM, Petty BA, et al. Optimal method of urgent decompression of the collecting system for obstruction and infection due to ureteral calculi. J Urol. 1998;160(4):1260–4.

- Dilauro MD, Dason S, Athreya S. Prophylactic balloon occlusion of internal iliac arteries in women with placenta accreta: literature review and analysis. Clin Radiol. 2012;67(6):515–20.
- Weltings S, Schout BMA, Roshani H, Kamphuis GM, Pelger RCM. Lessons from literature: nephrostomy versus double J ureteral catheterization in patients with obstructive urolithiasis-which method is superior? J Endourol. 2019;33(10):777–86.
- 29. Rosen A, Jayram G, Drazer M, Eggener SE. Global trends in testicular cancer incidence and mortality. Eur Urol. 2011;60(2):374–9.
- Bosl GJ, Goldman A, Lange PH, Vogelzang NJ, Fraley EE, Levitt SH, et al. Impact of delay in diagnosis on clinical stage of testicular cancer. Lancet. 1981;318(8253):970–3.
- Moul JW, Paulson DF, Dodge RK, Walther PJ. Delay in diagnosis and survival in testicular cancer: impact of effective therapy and changes during 18 years. J Urol. 1990;143(3):520–3.
- 32. Stephenson AJ, Russo P, Kaplinsky R, Sheinfeld J. Impact of unnecessary exploratory laparotomy on the treatment of patients with metastatic germ cell tumor. J Urol. 2004;171(4):1474–7.
- Arnold PM, Morgan CJ, Morantz RA, Eckard DA, Kepes JJ. Metastatic testicular cancer presenting as spinal cord compression: report of two cases. Surg Neurol. 2000;54(1):27–33.
- Robinson AG, Wei X, Karim S, Raphael MJ, Bedard PL, Booth CM. Venous thromboembolism during chemotherapy for testicular cancer: a population-based study. Clin Oncol (R Coll Radiol). 2020;32(10):e188–93.
- 35. Stephenson A, Eggener SE, Bass EB, Chelnick DM, Daneshmand S, Feldman D, et al. Diagnosis and treatment of early stage testicular cancer: AUA guideline. J Urol. 2019;202(2):272–81.
- 36. Gilligan T, Lin DW, Aggarwal R, Chism D, Cost N, Derweesh IH, et al. Testicular cancer, version 2.2020, NCCN clinical practice guidelines in oncology. J Natl Compr Cancer Netw. 2019;17(12):1529–54.
- Ciancio G, Soloway M. Renal cell carcinoma invading the hepatic veins. Cancer. 2001;92(7):1836–42.
- Psutka SP, Leibovich BC. Management of inferior vena cava tumor thrombus in locally advanced renal cell carcinoma. Ther Adv Urol. 2015;7(4):216–29.
- 39. Koster A, Dyke CM, Aldea G, Smedira NG, McCarthy HL, Aronson S, et al. Bivalirudin during cardiopulmonary bypass in patients with previous or acute heparin-induced thrombocytopenia and heparin antibodies: results of the CHOOSE-ON trial. Ann Thorac Surg. 2007;83(2):572–7.
- Labbate C, Hatogai K, Werntz R, Stadler WM, Steinberg GD, Eggener S, et al. Complete response of renal cell carcinoma vena cava tumor thrombus to neoadjuvant immunotherapy. J Immunother Cancer. 2019;7(1):1–6.
- American Cancer Society. Key statistics for penile cancer. Last medical review: 25 June 2018; last revised 8 Jan 2020. https://www. cancer.org/content/dam/CRC/PDF/Public/8783.00.pdf. Accessed 17 June 2020.
- Lucky MA, Rogers B, Parr NJ. Referrals into a dedicated British Penile Cancer Centre and sources of possible delay. Sex Transm Infect. 2009;85(7):527–30.
- Skeppner E, Andersson SO, Johansson J-E, Windahl T. Initial symptoms and delay in patients with penile carcinoma. Scand J Urol Nephrol. 2012;46(5):319–25.
- Stratton KL, Culkin DJ. A contemporary review of HPV and penile cancer. Oncology (Williston Park). 2016;30(3):245–9.
- Protzel C, Alcaraz A, Horenblas S, Pizzocaro G, Zlotta A, Hakenberg OW. Lymphadenectomy in the surgical management of penile cancer. Eur Urol. 2009;55(5):1075–88.
- 46. Nassiri N, Kogan S, Gibbon DG, Graham A. Multimodal endovascular palliation for femoral arterial blowout in the setting of metastatic vulvar carcinoma. Ann Vasc Surg. 2015;2(1):127.e11–5.

- Bhageerathy PS, Thomas A, Thomas V, Keshava SN, Peedicayil A. Femoral arterial blowout post groin recurrence in vulvar carcinoma - novel endovascular management. Indian J Surg Oncol. 2016;7(4):456–9.
- Siegel RL, Miller KD, Jemal A. Cancer statistics, 2019. CA Cancer J Clin. 2019;69(1):7–34.
- 49. Mak KS, Lee LK, Mak RH, Wang S, Pile-Spellman J, Abrahm JL, et al. Incidence and treatment patterns in hospitalizations for malignant spinal cord compression in the United States, 1998–2006. Int J Radiat Oncol Biol Phys. 2011;80(3):824–31.
- Crnalic S, Hildingsson C, Bergh A, Widmark A, Svensson O, Löfvenberg R. Early diagnosis and treatment is crucial for neurological recovery after surgery for metastatic spinal cord compression in prostate cancer. Acta Oncol. 2013;52(4):809–15.
- Huggins C, Hodges CV. Studies on prostatic cancer: I. the effect of castration, of estrogen and of androgen injection on serum phosphatases in metastatic carcinoma of the prostate. J Urol. 2002;167(2 Part 2):948–51.
- 52. Shore ND, Saad F, Cookson MS, George DJ, Saltzstein DR, Tutrone R, et al. Oral relugolix for androgen-deprivation therapy in advanced prostate cancer. N Engl J Med. 2020;382(23): 2187–96.
- O'Connor AR, Coakley FV, Meng MV, Eberhardt S. Imaging of retained surgical sponges in the abdomen and pelvis. Am J Roentgenol. 2003;180(2):481–9.

- 54. Shabsigh A, Korets R, Vora KC, Brooks CM, Cronin AM, Savage C, et al. Defining early morbidity of radical cystectomy for patients with bladder cancer using a standardized reporting methodology. Eur Urol. 2009;55(1):164–76.
- 55. Duivenvoorden WCM, Daneshmand S, Canter D, Lotan Y, Black PC, Abdi H, et al. Incidence, characteristics and implications of thromboembolic events in patients with muscle invasive urothelial carcinoma of the bladder undergoing neoadjuvant chemotherapy. J Urol. 2016;196(6):1627–33.
- Donahue TF, Bochner BH, Sfakianos JP, Kent M, Bernstein M, Hilton WM, et al. Risk factors for the development of parastomal hernia after radical cystectomy. J Urol. 2014;191(6):1708–13.
- 57. Jain S, Nyirenda T, Yates J, Munver R. Incidence of renal artery pseudoaneurysm following open and minimally invasive partial nephrectomy: a systematic review and comparative analysis. J Urol. 2013;189(5):1643–8.
- Ghoneim TP, Thornton RH, Solomon SB, Adamy A, Favaretto RL, Russo P. Selective arterial embolization for pseudoaneurysms and arteriovenous fistula of renal artery branches following partial nephrectomy. J Urol. 2011;185(6):2061–5.
- Potter SR, Kavoussi LR, Jackman SV. Management of diaphragmatic injury during laparoscopic nephrectomy. J Urol. 2001;165(4):1203–4.
- Chetta A, Aiello M, Olivieri D. Diaphragm paralysis after nephrectomy. Thorax. 2006;61(6):549.

Gynecology



Loise W. Wairiri, Tatjana Bozanovic, Tanja Pejovic, and Jenna M. Kahn

Case Study 1

A 63-year-old woman with stage IIIC high-grade undifferentiated ovarian cancer underwent extensive surgery followed by chemotherapy. However, the disease rapidly progressed and the course was complicated by ureteral obstruction that was relieved by bilateral percutaneous nephrostomy tubes, and deep venous thrombosis.

After the fifth cycle of carboplatin and Taxol, she presented to the emergency department with several days of worsening abdominal pain, distension, and nausea, and was found to be neutropenic with lactate level of 8. CT scan showed high-grade small bowel obstruction, closed loop in nature. The distal point was associated with a segment of edematous bowel, normal in caliber. A differential diagnosis of bowel obstruction due to malignancy or adhesions, and possible bowel ischemia seen on CT as edema, was made. Overall, the tumor disease burden seemed to have improved with no new disease sites.

The dilemma of emergent surgery for bowel obstruction and possible bowel ischemia and high mortality rate versus conservative management in the setting of progressive disease with a poor prognosis was discussed with the patient. A decision was made to proceed with surgery. During exploratory surgery, two distinct adhesive lesions that were tethering the bowel to the base of mesentery were found. They were lysed and the area of closed loop was released. There was no extensive disease in the abdomen, and only few 5 mm areas of tumor implants, ablated by chemotherapy, were at the small bowel mesentery. This case illustrates the

Department of Radiation Medicine, Oregon Health & Science University, Portland, OR, USA

e-mail: kahnje@ohsu.edu T. Bozanovic Clinic for Gynecology and Obstetrics, Clinical Centre of Serbia,

University of Belgrade, Belgrade, Serbia

T. Pejovic Department of Obstetrics and Gynecology, Oregon Health & Science University, Portland, OR, USA management dilemma frequently faced in the cases of bowel obstruction in patients with pelvic malignancies.

Case Study 2

An 80-year-old woman presents to the emergency department with abdominal pain, vomiting, and confusion. She has a history of ovarian cancer diagnosed 1 year ago s/p chemoradiation treatment. On examination; temp: 39.20c, HR: 110 bpm, BP: 85/60 mmHg, and RR: 26 breaths/min. Physical examination reveals a diffusely tender abdomen but no rebound tenderness, rigidity, or palpable masses. Discuss her plan of management and differential diagnosis.

Management This patient presents with septic shock, a complication arising from a possible differential diagnosis of bowel obstruction (malignant/non-malignant bowel obstruction) or pelvic infection. According to the Third International Consensus Definitions Task Force, sepsis is defined as "life threatening organ dysfunction due to a dysregulated host response to infection" [1].

Aggressive fluid resuscitation and empiric parenteral broad-spectrum antibiotics are required as initial management. Monitor vital signs and input/output as well as central venous pressure.

Sepsis is associated with organ failure. Comprehensive metabolic panel, liver function tests, and complete blood count are recommended laboratory studies. Leukocytosis can be seen in cases of pelvic infection, bowel obstruction, or sepsis.

Imaging may be done once the patient's condition is stable. Abdominopelvic CT with IV contrast is useful to rule out pelvic abscess, bowel perforation, or bowel ischemia. Pelvic abscess and bowel perforation are surgical emergencies, and early surgical management improves the outcome of the patient. It is crucial to evaluate the stage of the disease, patient's prognosis, and quality of life before considering surgical management as it is associated with high morbidity and mortality in terminal disease.

L. W. Wairiri · J. M. Kahn (🖂)

The patient's nutritional needs should also be considered. Evaluate the need for parenteral nutrition and consider the benefits vs the associated risks such as catheter infection, sepsis, and renal/hepatic failure.

It is important to have an appropriate and adequate pain management plan for the patient. The goal of care especially in palliative cases must include patient comfort.

Introduction

Gynecologic cancer patients can suffer unexpected emergencies at any stage of disease and treatment. Emergencies may be due to complications of the disease, toxic effects of treatments used in managing the malignancies including chemotherapy, surgical procedures, and radiotherapy, as well as multifactorial underlying causes. Different approaches in handling each emergency depend on the etiology. In this chapter, we discuss emergencies and complications frequently encountered in gynecologic oncology. We will also review the management of these emergencies taking into consideration that goals of care should be dependent on the patient's disease status and prognosis.

Acute Blood Loss

Hemorrhage is a frequent and potentially life-threatening complication in gynecologic oncology. The causes of bleeding are related to tumor invasion, tumor angiogenesis, clotting factor deficiencies arising from hepatic impairment, and systemic effects of cancer, such as chemotherapy and radiation therapy [2].

Primary or recurrent neoplasms of the uterus, cervix, vagina, or vulva may present with small bleeds or with massive life-threatening hemorrhage. Hemorrhage due to metastasis can also be of varying degrees with massive hemorrhage signifying the terminal event in late-stage disease. The most common cause of bleeding in gynecologic malignancies is advanced cervical cancer. Due to the varied incidence and stage of diagnosis of cervical cancer in the world, studies report incidence of vaginal bleeding ranging from 0.7% to 100%, with an attributable mortality of up to 6% in these patients [3]. Palliative treatment options for bleeding are used in different health-care settings depending on available resources.

Many gynecologic surgical procedures are also associated with significant bleeding due to the vascular tumor surgical sites. While most of the causes and areas of bleeding can be controlled by conventional management, it is important to mention presacral bleeding, which is not easily controlled by standard management and often fatal. However, presacral bleeding is uncommon and is seen in gynecologic oncology procedures such as pelvic exenteration and radical hysterectomy [4].

Other significant causes of acute blood loss in gynecologic malignancies include hemorrhagic cystitis and radiation proctitis, both of which are late toxic effects of pelvic radiation.

Radiation proctitis is radiation-induced inflammation of the rectal mucosa occurring as a result of pelvic radiation therapy. The rectum is particularly susceptible to radiation injury due to its proximity to the pelvic organs and fixed position in the pelvis [5]. Acute radiation proctitis, which is usually self-limiting, occurs in the first 3 months of radiation treatment and may present with mild rectal bleeding and tenesmus. Chronic radiation proctitis occurs 3 months after completion of radiation therapy and presents with more overt bleeding due to predominance of fragile telangiectatic vessels and a friable mucosa.

Hemorrhagic cystitis (HC) is a common adverse effect of chemotherapeutic agents, such as cyclophosphamide and isofosphamide used in bone and soft tissue sarcomas and also as immunosuppressive agents for hematopoietic cell transplantation. In gynecologic cancers, HC has been described as a late toxicity of pelvic radiation treatment. Depending on the severity and the cause of hemorrhagic cystitis, different grading systems are used (Table 32.1).

Radiation-induced hemorrhagic cystitis may present in the acute or chronic phase. Acute radiation-induced HC occurs in 10% of gynecologic cancers treated with pelvic radiation [6] and presents with bladder pain that resolves ~ 4-6 weeks after the last fraction of radiotherapy. It is prudent to rule out a urinary tract infection, which presents with similar symptoms. The chronic phase of HC occurs 2-10 years post radiotherapy, and presents with bladder irritation and hematuria. This stage is irreversible. It is important to rule out bladder cancer, which can occur either as a primary disease or secondary late complication of pelvic radiation. Therefore, urinary examination, CT urogram, and cystoscopy with biopsy of lesions should be done to differentiate malignancy from chronic inflammation. Confirmation of chronic radiation inflammation is achieved by demonstrating that the inflammatory lesion occurred in the same site of radiation as per the dosimetry plan (Table 32.1) [7].

Severe hemorrhagic cystitis, though not common, has significantly high mortality despite vigorous treatment. A large study conducted to study the incidence, severity, timing, clin-

 Table 32.1
 Grading of hemorrhagic cystitis [8]

Grade	Features
0	No symptoms
1	Microscopic hematuria
2	Macroscopic hematuria
3	Macroscopic hematuria with small clots
4	Massive hematuria requiring clot evacuation

ical management, and outcome for patients who developed hemorrhagic cystitis following pelvic therapy for Federation of Gynecology and Obstetrics (FIGO) stage IB cancer of the cervix showed the incidence of HC to be 6.5%, with grade 3 occurring in 18% of the HC cases. After actuarial analysis, they demonstrated an increase in incidence with increasing duration from the last pelvic radiotherapy treatment [9]. Clinicians should be aware and recognize hemorrhagic cystitis emergencies that occur as late effects of radiation toxicity.

Management

Immediate assessment of hemodynamic stability is important for timely resuscitation to be initiated. Adequate volume replacement with crystalloids, followed by blood products, should be administered with frequent reassessment and monitoring of the patient's clinical status and vital signs. In patients with cardiorespiratory impairment, titration against central venous pressure may be necessary. Insert a Foley catheter to monitor output and also to avoid urinary retention by inserting prior to vaginal packing. Management depends on individual needs including platelets and clotting factors replacement. Depending on the cause, options to control bleeding include non-invasive and invasive methods.

When the cause of bleeding is due to advanced cervical cancer, palliative interventions have been used with no single intervention demonstrated to be more superior. Resources play a major role regarding the choice of intervention used in the management of bleeding [3]. Non-invasive techniques are utilized first to control hemorrhage, and when these fail, more invasive methods are applied.

Non-invasive Bleeding Management Vaginal packing is a simple initial measure to control vaginal bleeding, a common presentation in advanced cervical cancer. Sedation or short-acting general anesthesia may be utilized to ensure patient comfort. The source of bleeding is visualized and by use of speculum, the vaginal fornices are packed tightly to maintain an even pressure on the bleeding vessels. Measures to enhance the effectiveness of the packing include decreasing patient mobility, elevating the foot of the bed, and using hemostatic agents (oral, local, or parenteral). Tranexamic acid is an example of a parenteral agent that increases clotting by inhibiting plasmin in the clotting cascade. Topical/ locally acting hemostatic agents can be used, including Monsel solution, formalin, and Moh's paste [3, 10].

Monsel solution(ferric subsulfate) can be applied directly on the bleeding site on the cervix or soaked in the vaginal pack and causes agglutination of protein resulting in sealing of the blood vessels [3]. Although Monsel solution has been reported in numerous studies as an excellent hemostatic agent, great care should be observed that minimal amounts of the solution are used. Due to its action, it can induce fullthickness necrosis, perforation, and life-threatening peritonitis when exposed to the peritoneal cavity [11]. Moh's paste has been used as a safe and effective method in cases of vaginal bleeding due to advanced cervical cancer though clinical trials are warranted for adoption of widespread use. It works by releasing zinc ions when in contact with cervical tumors and precipitates wound proteins to enhance hemostasis [12]. Formalin has been used in limited resource settings to enhance the effectiveness of the vaginal pack by acting as a chemical cauterization agent inducing contact coagulative tissue necrosis of the neovasculature to stop bleeders [10]. Formalin is also used to control bleeding in radiation proctitis with documented success rates of up to 75%. Caution must be taken to prevent damage to normal rectal mucosa, which can cause anal pain, strictures, fistulas, and stool incontinence [5, 13]. (Detailed management of radiation proctitis is covered in another chapter of the book.)

When using a vaginal pack to control bleeding, oral broad-spectrum antibiotics and metronidazole are required to treat underlying infections that are common in necrotic, bleeding tumors [10].

Another non-invasive method to control bleeding in gynecologic tumors is hemostatic radiotherapy.

Hemostatic radiotherapy (RT) is effective for emergent palliation of bleeding tumors with lower recurrence of re-bleed and minimal treatment burden for patients with advanced disease. Short-course hypo-fractionated RT is effective and well-tolerated for vaginal bleeding control in 93.8% of patients with advanced cervical cancer, with an additional benefit of pelvic pain control in 66.7% of the patients [14]. Comparable results have been reported of symptomatic relief of vaginal bleeding in 85–100% of patients when treated with two or three 10 Gy fractions of pelvic radiation therapy [15]. More evidence also supports palliative-dose radiotherapy at standard curative treatment fraction of 1.8–2.0 Gy in a single-dose fraction effective for vaginal bleeding in cervical cancer [3, 10].

Invasive Bleeding Management Invasive methods such as interventional radiology and surgery are reserved for use in cases of uncontrolled bleeding in carefully selected patients due to associated morbidity. Uterine artery embolization using metal coils, foam, or glue-like agents is done by the interventional radiology (IR) team, and when IR is not available, laparotomy with ligation of vessels (uterine artery and divisions of the hypogastric arteries) serve as an alternative [4].

Hemorrhagic Cystitis Treatment Radiation-induced hemorrhagic cystitis poses challenges in management due to the ischemic nature of the disease. There are no existing guidelines and clinical trials for treatment. Resuscitation with fluids or blood products should be used for hemodynamically unstable patients while carefully monitoring the vital signs and central venous pressure.

Management depends on the severity and grading of HC. Various options are used to alleviate symptoms, including steroids, vitamin E, trypsin, intravesical therapy with formalin and hyaluronic acid for clot evacuation and chemical cauterization, and hyperbaric oxygen with varying degrees of success. Visualization of the fragile telangiectatic vessels by cystoscopy and selective embolization of the bleeders can be used after non-invasive steps are ineffective. Surgery may be considered for refractory cases as a last treatment option. Strict dose constraints to the bladder and tailoring the field are employed to reduce the incidence of hematuria [7, 8, 16].

Summary: Management of Hemorrhagic Cystitis (Fig. 32.1) [7]

Pelvic Infections

Radical gynecologic oncology surgery is associated with a high rate of postoperative complications. Gynecologic procedures increase the risk of pelvic infections due to migration of potential microorganisms from the skin, vagina, and endocervix [17].

Additionally, inefficient defense mechanisms as seen in gynecologic cancer patients, impaired venous or lymphatic

circulation after radical hysterectomy or radical vulvectomy [18], and a history of radiation therapy further increase the risks of infection [17].

Pelvic infections in gynecologic cancer patients included in this chapter are vaginal cuff cellulitis, pelvic cellulitis, and pelvic abscesses. Vaginal cuff cellulitis is an infection of the superficial tissues at the vaginal surgical margin, while pelvic cellulitis involves deep soft tissue at the incision.

Prior to routine antibiotic prophylaxis, pelvic infection rates post-hysterectomy were up to 33% [16]; however, widespread implementation of routine antibiotic prophylaxis has led to a significant decrease of pelvic infection rates to 2.7% [18, 19]. Other studies report that pelvic abscesses occur in 3.9% of patients within 30 days after surgery and 6.6% of patients as a late complication in post-trachelectomy for cervical cancer [20, 21].

Infections are often polymicrobial in nature, composed of both endogenous non-pathogenic and pathogenic bacteria. Frequently isolated pathogen includes Gram-negative bacteria, enterococci, group B streptococci, and anaerobes.

Typically, patients with pelvic abscess present with pelvic pain, fever, tachycardia, and tachypnea. On physical exam, the abdomen/pelvis is diffusely tender and a fluctuant mass may be palpable. Vaginal cuff cellulitis presents as acute onset of fever, chills, and macular erythema, with swelling over the lower abdominal wall and inguinal region, that may extend to the proximal thigh. The adnexa and parametrium are non-tender to physical exam. Pelvic cellulitis presents with fever, vague abdominal pain, and anorexia. Physical



Diagnosis	Treatment regimen	Notes
Vaginal cuff cellulitis	Amoxicillin/clavulanate 875/125 mg every 12 h or Ciprofloxacin 500 mg every 12 h with metronidazole 500 mg every 12 h or TMP-SMX DS every 12 h with metronidazole 500 mg every 12 h	Empiric broad-spectrum antibiotics outpatient management with close follow-up
Pelvic cellulitis	Clindamycin 900 mg every 8 h <i>or</i> metronidazole 500 mg every 12 h <i>plus</i> Penicillin five million units every 6 h <i>or</i> ampicillin 2 g every 6 h <i>plus</i> Gentamicin 5 mg/kg every 24 h In case of <i>renal impairment substitute gentamicin for aztreonam 2 g</i> <i>every 8 h</i>	Empirical inpatient parenteral broad-spectrum antibiotics, administered intravenously until patient is afebrile for 24–48 h and can then be discharged with oral antibiotic regimen for 14 days with close follow-up
Pelvic abscess	 Metronidazole 500 mg every 12 h plus Ceftriaxone 2 g every 24 h^a or Piperacillin-tazobactam 3.375 g every 6 h or Aztreonam <i>plus</i> clindamycin or carbapenem^b 	Empiric parenteral broad-spectrum antibiotics continued until patient is afebrile for 48–72 h and clinical improvement is seen. Choice of oral antibiotics should depend on culture and sensitivity results

Table 32.2 Choice of antibiotic therapy for pelvic infections

^aRecommended first line

^bIn case of penicillin or cephalosporin allergy

examination of pelvic cellulitis will reveal tenderness with absence of masses unlike pelvic abscess [18, 22].

Laboratory findings include leukocytosis, elevated erythrocyte sedimentation rate, and elevated C-reactive protein. Fluid collections in pelvic abscess can be visualized by ultrasonography, computed tomography (CT), or magnetic resonance imaging (MRI).

Management

Management depends on the clinical status of the patient and the characteristics of the infection. Initial assessment includes evaluation of the hemodynamic stability of the patient. Hemodynamically unstable patients are likely to have a ruptured abscess, which is life-threatening and requires prompt emergent surgical intervention. In addition, these patients require aggressive fluid resuscitation and empiric parenteral broad-spectrum antibiotics [22].

Vaginal cuff cellulitis is managed in the outpatient setting with broad-spectrum antibiotics and close follow-up to monitor treatment efficacy. Pelvic cellulitis and pelvic abscess require in-hospital management. Surgical abscesses require drainage if; the pelvic abscess is >8 cm, there is no adequate response/ clinical deterioration to antibiotic treatment for 24–48 h, and progressively increasing abscess size [23]. There is sufficient evidence that suggests early drainage of all abscesses combined with broad-spectrum antibiotics is safe and improves clinical outcomes with reduced mean hospital stays [24, 25]. However, due to associated morbidity in these patients, the clinician should evaluate the increased risks versus potential benefits.

The suggested regimens (Table 32.2) are based on broad consensus rather than clinical trials. If the patient does not respond 48–72 h on treatment, consultation with an infectious disease specialist is warranted.

Gastrointestinal Emergencies in Gynecologic Cancer Patients

Malignant Bowel Obstruction

Acute intestinal obstruction is a surgical emergency that is often associated with pelvic tumors. It is a relatively common complication of advanced gynecologic malignancies with reported rates of 6% to 50% in patients with advanced or recurrent ovarian cancer and ~5% in cervical cancer [26, 27]. Bowel obstruction can be classified according to etiology as malignant bowel obstruction (MBO) or non-malignant bowel obstruction [26].

The causes of MBO in gynecologic cancer include peritoneal carcinomatosis, extrinsic bowel compression by tumor, adhesions, enlarged pelvic lymph nodes, tumor infiltration of the mesentery, bowel muscle or nerves, edema of the bowel wall, and post-radiation changes [28, 29].

In gynecologic cancer patients, MBO has a relatively poor prognosis with some studies reporting 3-month median survival rates. MBO has often been cited as a pre-terminal event for majority of the patients regardless of treatment. Therefore, the goal is palliative management for comfort and improving their quality of life [30]. Diagnosis of intestinal obstruction is made by presenting clinical symptoms and diagnostic imaging. Proximal bowel obstruction presents more abruptly with abdominal pain, nausea, and vomiting. Repeated vomiting may lead to metabolic alkalosis and hypokalemia. Distal bowel obstruction is more insidious and progressive in onset, presenting with prominent abdominal distension, abdominal pain, and constipation.

Plain abdominal radiography is the initial imaging of choice and demonstrates features of obstruction that include dilated loops of bowel with multiple air fluid levels [31]. Abdominopelvic computed tomography (CT) gives more clinical information, which is relevant in planning management. It determines presence and level of obstruction and complications of bowel obstruction such as ischemia or perforation of involved bowel, both of which are surgical emergencies [32].

Laboratory studies are also useful as they evaluate presence of complications such as metabolic derangements (acidosis, hyperkalemia) and leukocytosis which can be suggestive of infection.

Management The aim of management in gynecologic cancer patients with MBO is symptom control and maintain the patient's quality of life due to the associated high comorbidities and low median survival rates.

Therapy starts with conservative management, which includes restoration of fluid and electrolyte balance, alternatives for feeding, restriction of medication that have paralytic effect on the intestines, and nasogastric tube placement for decompression with stimulation of intestinal passage for distal obstructions [33]. This conservative regimen will keep the patient in optimal condition and allow time for to identify the origin of the obstruction, stage of the malignant disease, and obtain multidisciplinary evaluation. Conservative treatment provides time for diagnostic procedures as well as to see if the obstruction will resolve spontaneously, but no longer than 3–7 days [34, 35]. After this period of time, decisions have to be made either for surgery or refraining from intervention and providing symptomatic and supportive care.

There are contradicting clinical and ethical accounts supporting and against parenteral nutrition for patients with advanced malignancies receiving parenteral care. Patients with MBO have limited/no oral intake, and parenteral nutrition (PN) is a way of providing macro- and micronutrients, fluids, and electrolytes. There is still very low evidence whether parenteral nutrition plays a role in improving survival and quality of life in people with malignant bowel obstruction [36]. TPN has been shown to be of benefit only to improve immediate survival in patients who may otherwise die of starvation than the malignancy [37]. In retrospective studies, PN has been associated with improved survival in patients with a Karnofsky Performance Score (KPS) of greater than 50 [38]. However, the benefits of home PN in patients with inoperable malignant MBO are uncertain and must be balanced against risk of central line infections, thrombosis, electrolyte abnormalities, and increased cost with no associated increase in mean survival for the patient. Consider PN when the risk of starvation or malnutrition is higher than that of disease progression. The European Society for Clinical Nutrition and Metabolism (ESPEN) recommends PN should be offered if the expected survival with tumor progression is longer than 2–3 months [38].

When the cause of obstruction is benign, laparotomy should be performed for adhesiolysis or bowel resection [39]. In the case of radiation enteritis, it is important to resect the entire diseased bowel segment to reduce recurrence, postoperative complications, and mortality [40]. In malignant obstruction, surgical interventions such as bowel resection, bypass, or ileostomy may seem to provide good palliation by reduction of symptoms and obstruction recurrence in progressive disease, but these interventions depend on the extent of the disease in the individual patient [41]. Palliative surgery offers limited benefits in improving quality of life and survival. Family members should be involved in critical discussions prior to undertaking such drastic measures [36]. Surgery in patients with peritoneal carcinomatosis is related with a 30-day mortality of 21–40% and high recurrence rates [42].

Gastrointestinal Perforation and Fistula

Gastrointestinal (GI) perforation and fistulas are potentially life-threatening events that can develop in gynecologic cancer patients. Patients are prone to bowel injury of various forms, including complications arising from tumor invasion, pelvic radiation, and chemotherapy. Tumor invasion of the bowel is common in advanced ovarian, fallopian tube, and primary peritoneal cancers. Pelvic radiation, frequently used in advanced cervical cancer treatments, is a known cause of radiation-related bowel toxicities.

Bevacizumab, an antiangiogenic drug that was approved in 2014 for patients with advanced cancer, has significantly improved overall survival when used in combination with other chemotherapy agents. However, it is associated with clinically significant rates of GI fistula: grade 2 and grade 3 (requiring intervention), at rates of 5% and 3%, respectively. The risk of developing fistulas is increased by additional clinical factors such as concurrent tobacco use, pre-existing hypertension, pelvic tumor (as an independent risk factor), and pelvic radiation [43]. Gastrointestinal perforation in bevacizumab is a serious but uncommon adverse effect that has been well documented in phase III clinical trials, as well as subsequent surveillance trials, with reported incidence rates of 0-11% [44].

Paclitaxel has also been associated with GI perforations, and it is believed it is caused by cellular necrosis induced
by mitotic arrest of GI epithelium. The exact incidence is unknown. However, while it is infrequent, it has a high mortality rate of 57%. Therefore quick diagnosis and a high index of suspicion in a patient presenting with abdominal pain and neutropenic fever following paclitaxel chemotherapy are warranted [45, 46, 68].

Presentation of bowel perforation and fistulas may be asymptomatic. Non-specific signs also include diffuse abdominal pain and tenderness, fever, and tachycardia.

Sepsis is a life-threatening complication of GI perforation and presents with hemodynamic instability, altered mental status, hypothermia/hyperthermia, and organ dysfunction, including acute respiratory distress syndrome, acute renal injury, and disseminated intravascular coagulopathy. Elevation of inflammatory markers (C-reactive protein, serum amylase) may be present, but is non-specific.

Patients with a suspected bowel perforation should be evaluated by abdominopelvic CT, which has higher sensitivity than plain films and can demonstrate small amounts of extraluminal gas.

Management Initial management should include an assessment of hemodynamic stability and monitoring of electrolytes. Resuscitation with intravenous fluid therapy is the next step in management for hemodynamically unstable patients.

Bowel injury in patients with advanced gynecologic malignancies tends to have a poor prognosis independent of the bowel injury.

Depending on the clinical status of the patient, appropriate conservative or definitive surgical management should be instituted. Non-invasive interventions include broad-spectrum antibiotics and, depending on the level of perforation, appropriate drainage tubes. Patients with suspected perforation whose clinical condition is unstable require immediate surgical repair or diversion. Factors that should be considered before surgical management are the patient's comorbidities, general clinical health condition, and overall prognosis. Extent/stage of the malignancy at the time of bowel perforation is the main prognostic indicator regardless of the type of management approach. Patients with advanced disease may be appropriate candidates for less invasive management [47].

Genitourinary Emergencies in Gynecologic Cancer Patients

Vaginal Cuff Dehiscence

Vaginal cuff dehiscence is a rare, but devastating, complication of hysterectomies (common procedures done in gynecologic oncology) and is associated with high morbidity requiring urgent management. A large study review of hysterectomies reported an annual cumulative incidence rate of vaginal dehiscence of ~ 0.14% of total abdominal and vaginal hysterectomies. Various studies report a similar low incidence. There is a slightly higher incidence reported in laparoscopicand robotic-assisted total hysterectomies [48, 49].

Patients with vaginal cuff dehiscence may present with abdominal pain, serous vaginal discharge, vaginal bleeding, and vaginal bulge, as well as with more severe complications, such as peritonitis and bowel evisceration.

Vaginal cuff dehiscence is a clinical diagnosis. A comprehensive physical and vaginal examination is adequate to make a definitive diagnosis. Recommended laboratory studies include a metabolic panel and complete blood count. Leukocytosis can signify complications such as infection, peritonitis, and bowel ischemia/injury. Imaging such as abdominopelvic CT may be used to rule out complications and comorbidities such as pelvic abscess, bowel injury, and hematoma [50].

Management The clinical status of the patient is an important aspect to consider in management of vaginal cuff dehiscence.

Initial management includes resuscitation of the patient with fluids and/or blood depending on the hemodynamic status of the patient. Broad-spectrum antibiotics that adequately cover for Gram-negative and anaerobic organisms (ampicillin, gentamicin, and clindamycin) should also be administered [50].

Expectant or definitive management is dependent on the patient's clinical status, size of the defect, and presence of bowel evisceration. Peritonitis and bowel evisceration are surgical emergencies. If bowel evisceration is present, while waiting to take the patient to the operating room, insert a Foley catheter, position the patient in Trendelenburg, and cover the bowel with a warm moist towel, frequently irrigating the structures with warm normal saline. The surgical approach relies on the surgeon's experience and status of bowel viability. Intraoperatively, in the absence of infections and with signs of viable bowel, the prolapsed bowel can be gently reduced to the peritoneal cavity transvaginally. Other surgical approaches include abdominal, laparoscopy, or a combination of both. Debridement of necrosed tissue, dissection, and bowel resection are common intraoperative procedures for surgical management of vaginal cuff dehiscence [51, 52].

Urinary Tract Obstruction

Urinary tract obstruction is an important cause of renal failure in gynecologic cancer patients, especially in advanced cervical cancer. Obstruction can be caused either by extraluminal compression on ureters or direct infiltration of tumor into ureters. Benign causes for acute obstruction of urinary tract in cancer patients can be fibrosis or pelvic inflammatory disease after surgery, catheter-induced edema, or strictures after radiation therapy [53].

Tumors causing obstruction include cervical cancer (most common), ovarian cancer, and other pelvic tumors. Large pelvic masses such as ovarian cancer can cause bilateral ure-teric obstruction [54]. When bilateral obstruction develops, it leads to anuria and renal failure, with progressive rise in serum creatinine [55].

Commonly, obstruction of the urinary tract leads to hydronephrosis. Ultrasound of the abdomen, cystoscopy, retrograde ureteric investigations, and CT scan are helpful diagnostic options in assessment of the etiology and extent of the obstruction.

Management The basic principle of management is decompression of ureters because it preserves renal function [56]. This can be accomplished by percutaneous nephrostomy or cystos-copy and retrograde placement of an internal ureteric stent [56, 57, 69]. Percutaneous nephrostomy is a temporary measure used for patients with undiagnosed malignancy or in patients with cervical cancer who have available treatment modalities and a good chance of treatment response. Ureteric stent insertion is reserved for patients with advanced malignancy for palliative relief of the obstruction and associated symptoms.

Pain

Acute pain is the most common symptom in gynecologic malignancies and is the leading cause of emergency department visits [58, 59]. Several retrospective studies have shown

that gynecologic cancer patients have higher rates of moderate to severe pain and higher rates of opioid use as compared to patients with other solid tumors [60].

In advanced cancers, pain can be caused by involvement of the regional nerves, pelvic muscles, and bones. Pelvic tumors can also encroach the presacral area causing visceral pain. The mechanisms of pain are not completely understood and are thought to include tissue destruction or stimulation of cytokine secretion [61].

Pain management depends on the mechanism and location of pain. The American Pain Society recommends the prompt treatment of patients with acute pain with numerous trials advocating multimodal therapy [61]. For patients with acute uncontrolled pain not responding to oral or IV analgesics, techniques such as epidural opioids and neurolysis can be used [10]. The National Comprehensive Cancer Network (NCCN) also confirms the use of opioids for severe to moderate pain and non-opioids as adjuncts. Neuropathic pain is frequently resistant to opioids and adjuncts such as tricyclic antidepressants (TCAs), selective serotonin receptor inhibitors (SSRIs), and gabapentin can be considered although there are limited clinical trials attesting to their efficacy [62].

The revised World Health Organization (WHO) analgesic ladder (Fig. 32.2) describes a simple four-step bidirectional approach for pain management, with opioids being the cornerstone of treatment for active cancer pain [63]. This approach should be used with caution due to the risk of opioid dependency and misuse. The American Society of Clinical Oncology (ASCO) policy statement on the use of opioids in the cancer population states that access to opioids must be assured, and law and regulations intended to address abuse and overdose should be crafted to avoid creating impediments to this treatment [64]. The European



Association for Palliative Care advises that the skilled use of opioids is crucial to the relief of cancer pain [65].

Palliative radiation therapy is the standard for management of pain in bone metastasis. Radiation therapy (RT) has been proven to provide effective palliation of painful bone metastases with few adverse effects. The American Society for Radiation Oncology (ASTRO) has strong recommendations based on high quality evidence for the use of palliative RT for pain management. Updated reviews show equivalent pain relief for single fractions for patients with previously unirradiated bone metastases, with no increased risk for pathologic fractions, as with fractionated therapy. This is desirable and convenient for patients with advanced disease and limited life expectancy [66].

Conclusion

Emergencies in patients with a history of gynecologic malignancy can occur at any time during the course of cancer disease. Specific features of emergency presentations in these patients require knowledge of the patient's cancer history and medical knowledge of principles that should be applied in emergency situations. Therapy is individually tailored and depends on the underlying cause for the emergency, stage of the malignant disease if still present, previous cancer treatments, and immunological and general condition of the patient. Prompt therapy is crucial and diagnostic imaging should not delay initial management. Computerized tomography imaging guidelines and early gynecologic oncology consultations have significantly decreased CT utilization rates in the ED, therefore avoiding unnecessary imaging [67].

References

- Seymour CW, Liu VX, Iwashyna TJ, Brunkhorst FM, Rea TD, Scherag A, et al. Assessment of clinical criteria for sepsis: for the Third International Consensus Definitions for Sepsis and Septic Shock (Sepsis-3). JAMA. 2016;315(8):762–74. Erratum in: JAMA. 2016;315(20):2237.
- Johnstone C, Rich SE. Bleeding in cancer patients and its treatment: a review. Ann Palliat Med. 2018;7(2):265–73.
- Eleje GU, Eke AC, Igberase GO, Igwegbe AO, Eleje LI. Palliative interventions for controlling vaginal bleeding in advanced cervical cancer. Cochrane Database Syst Rev. 2019;3(3):CD011000.
- Wydra D, Emerich J, Ciach K, Marciniak A. Management of massive presacral pelvic bleeding in patients with gynecological malignances: review of the literature. Eur J Gynaecol Oncol. 2005;26(2):151–4.
- Tabaja L, Sidani SM. Management of Radiation Proctitis. Dig Dis Sci. 2018;63(9):2180–8.
- Powell ME. Modern radiotherapy and cervical cancer. Int J Gynecol Cancer. 2010;20(11 Suppl 2):S49–51.

- Liem X, Saad F, Delouya G. A practical approach to the management of radiation-induced hemorrhagic cystitis. Drugs. 2015;75(13):1471–82. https://doi.org/10.1007/s40265-015-0443-5.
- Manikandan R, Kumar S, Dorairajan LN. Hemorrhagic cystitis: a challenge to the urologist. Indian J Urol. 2010;26(2):159–66.
- 9. Levenback C, Eifel PJ, Burke TW, Morris M, Gershenson DM. Hemorrhagic cystitis following radiotherapy for stage ib cancer of the cervix. Gynecol Oncol. 1994;55(2):206–10.
- Mishra K. Gynaecological malignancies from palliative care perspective. Indian J Palliat Care. 2011;17(Suppl):S45–51.
- Shuhaiber JH, Lipnick S, Teresi M, Molinari M, Ryoo J, Cintron J. More on Monsel's solution. Surgery. 2005;137(2):263–4.
- Yanazume S, Douzono H, Yanazume Y, Iio K, Douchi T. New hemostatic method using Mohs' paste for fatal genital bleeding in advanced cervical cancer. Gynecol Oncol Case Rep. 2013;4:47–9.
- Ma TH, Yuan ZX, Zhong QH, Wang HM, Qin QY, Chen XX, et al. Formalin irrigation for hemorrhagic chronic radiation proctitis. World J Gastroenterol. 2015;21(12):3593–8.
- Kim HD, Lee HJ, Ki KY, Nam HJ, Kim TW, Jeon SH, et al. Shortcourse palliative radiotherapy for uterine cervical cancer. Radiat Oncol J. 2013;31(4):216–21.
- Yan J, Milosevic M, Fyles A, Manchul L, Kelly V, Levin W. A hypofractionated radiotherapy regimen (0-7-21) for advanced gynaecological cancer patients. Clin Oncol. 2011;23(7):476–81.
- Denton AS, Clarke NW, Maher EJ. Non-surgical interventions for late radiation cystitis in patients who have received radical radiotherapy to the pelvis. Cochrane Database Syst Rev. 2002;2002(3):CD001773.
- Lachiewicz MP, Moulton LJ, Jaiyeoba O. Pelvic surgical site infections in gynecologic surgery. Infect Dis Obstet Gynecol. 2015;2015:614950.
- Santos MB, Soares RF, Basto F. Streptococcal sex syndrome: a curious association between sex and cellulitis. Gynecol Oncol Rep. 2018;26:11–3.
- Lake AG, McPencow AM, Dick-Biascoechea MA, Martin DK, Erekson EA. Surgical site infection after hysterectomy. Am J Obstet Gynecol. 2013;209(5):490.e1.
- Fagotti A, Costantini B, Fanfani F, Vizzielli G, Rossitto C, Lecca A, et al. Risk of postoperative pelvic abscess in major gynecologic oncology surgery: one-year single-institution experience. Ann Surg Oncol. 2010;17(9):2452–8.
- Okugawa K, Sonoda K, Ohgami T, Yasunaga M, Kaneki E, Yahata H, et al. Pelvic abscess: a late complication of abdominal trachelectomy for cervical cancer. J Obstet Gynaecol Res. 2019;45(2):412–6.
- Mazuski JE, Tessier JM, May AK, Sawyer RG, Nadler EP, Rosengart MR, et al. The Surgical Infection Society Revised Guidelines on the Management of Intra-Abdominal Infection. Surg Infect (Larchmt). 2017;18(1). doi:10.1089/sur.2016.261
- 23. Dewitt J, Reining A, Allsworth JE, Peipert JF. Tuboovarian abscesses: is size associated with duration of hospitalization & complications? Obstet Gynecol Int. 2010;2010:847041.
- Perez-Medina T, Huertas MA, Bajo JM. Early ultrasound-guided transvaginal drainage of tubo-ovarian abscesses: a randomized study. Ultrasound Obstet Gynecol. 1996;7(6):435–8.
- Gjelland K, Ekerhovd E, Granberg S. Transvaginal ultrasoundguided aspiration for treatment of tubo-ovarian abscess: a study of 302 cases. Am J Obstet Gynecol. 2005;193(4):1323–30.
- Caceres A, Zhou Q, Iasonos A, Gerdes H, Chi DS, Barakat RR. Colorectal stents for palliation of large-bowel obstructions in recurrent gynecologic cancer: an updated series. Gynecol Oncol. 2008;108(3):482–5.
- Lee YC, Jivraj N, O'Brien C, Chawla T, Shlomovitz E, Buchanan S, et al. Malignant bowel obstruction in advanced gynecologic cancers: an updated review from a multidisciplinary perspective. Obstet Gynecol Int. 2018;2018:1867238.

- Soo KC, Davidson T, Parker M, Paterson I, Paterson A. Intestinal obstruction in patients with gynaecological malignancies. Ann Acad Med Singap. 1988;17(1):72–5.
- Furnes B, Svensen R, Helland H, Ovrebo K. Challenges and outcome of surgery for bowel obstruction in women with gynaecologic cancer. Int J Surg. 2016;27:158–64.
- Kucukmetin A, Naik R, Galaal K, Bryant A, Dickinson HO. Palliative surgery versus medical management for bowel obstruction in ovarian cancer. Cochrane Database Syst Rev. 2010;2010(7):CD007792.
- Thompson WM, Kilani RK, Smith BB, Thomas J, Jaffe TA, Delong DM, et al. Accuracy of abdominal radiography in acute smallbowel obstruction: does reviewer experience matter? AJR Am J Roentgenol. 2007;188(3):W233–8.
- 32. Maung AA, Johnson DC, Piper GL, Barbosa RR, Rowell SE, Bokhari F, et al. Eastern Association for the Surgery of Trauma. Evaluation and management of small-bowel obstruction: an Eastern Association for the Surgery of Trauma practice management guideline. J Trauma Acute Care Surg. 2012;73(5 Suppl 4):S362–9.
- Bosscher MR, van Leeuwen BL, Hoekstra HJ. Surgical emergencies in oncology. Cancer Treat Rev. 2014;40(8):1028–36.
- Miller G, Boman J, Shrier I, Gordon PH. Readmission for smallbowel obstruction in the early postoperative period: etiology and outcome. Can J Surg. 2002;45(4):255–8.
- Turnbull ADM, Guerra J, Starnes HF. Results of surgery for obstructing carcinomatosis of gastrointestinal, pancreatic, or biliary origin. J Clin Oncol. 1989;7(3):381–6.
- 36. Sowerbutts AM, Lal S, Sremanakova J, Clamp A, Todd C, Jayson GC, Teubner A, Raftery AM, Sutton EJ, Hardy L, Burden S. Home parenteral nutrition for people with inoperable malignant bowel obstruction. Cochrane Database Syst Rev. 2018;8(8):CD012812.
- Soriano A, Davis MP. Malignant bowel obstruction: individualized treatment near the end of life. Cleve Clin J Med. 2011;78(3):197–206.
- Plana A, Izquierdo FJ, Schuitevoerder D, Lambert LA, Micic D, Deneve JL, et al. The Chicago Consensus on peritoneal surface malignancies: palliative care considerations. Cancer. 2020;126(11):2571–6.
- Mirensky TL, Schuster KM, Ali UA, Reddy V, Schwartz PE, Longo WE. Outcomes of small bowel obstruction in patients with previous gynecologic malignancies. Am J Surg. 2012;203(4):472–9.
- Chiarugi M, Galatioto C, Panicucci S, Scassa F, Zocco G, Seccia M. Oncologic colon cancer resection in emergency: are we doing enough? Surg Oncol. 2007;16(Suppl 1):S73–7.
- 41. Abbas SM, Merrie AEH. Resection of peritoneal metastases causing malignant small bowel obstruction. World J Surg Oncol. 2007;5:122.
- 42. Ripamonti CI, Easson AM, Gerdes H. Management of malignant bowel obstruction. Eur J Cancer. 2008;44(8):1105–15.
- 43. Tewari KS, Sill MW, Penson RT, Huang H, Ramondetta LM, Landrum LM, et al. Bevacizumab for advanced cervical cancer: final overall survival and adverse event analysis of a randomised, controlled, open-label, phase 3 trial (Gynecologic Oncology Group 240). Lancet. 2017;390(10103):1654–63.
- 44. Burger RA, Brady MF, Bookman MA, Monk BJ, Walker JL, Homesley HD, et al. Risk factors for GI adverse events in a phase III randomized trial of bevacizumab in first-line therapy of advanced ovarian cancer: a gynecologic oncology group study. J Clin Oncol. 2014;32(12):1210–7.
- Rose PG, Piver MS. Intestinal perforation secondary to paclitaxel. Gynecol Oncol. 1995;57(2):270–2.

- 46. Seewaldt VL, Cain JM, Goff BA, Tamimi H, Greer B, Figge D. A retrospective review of paclitaxel-associated gastrointestinal necrosis in patients with epithelial ovarian cancer. Gynecol Oncol. 1997;67(2):137–40.
- Wydra D, Emerich J, Sawicki S, Ciach K, Marciniak A. Major complications following exenteration in cases of pelvic malignancy: a 10-year experience. World J Gastroenterol. 2006;12(7):1115–9.
- Hur HC, Guido RS, Mansuria SM, Hacker MR, Sanfilippo JS, Lee TT. Incidence and patient characteristics of vaginal cuff dehiscence after different modes of hysterectomies. J Minim Invasive Gynecol. 2007;14(3):311–7.
- Fuchs Weizman N, Einarsson JI, Wang KC, Vitonis AF, Cohen SL. Vaginal cuff dehiscence: risk factors and associated morbidities. JSLS. 2015;19(2):e2013.00351.
- Matthews CA, Kenton K. Treatment of vaginal cuff evisceration. Obstet Gynecol. 2014;124(4):705–8.
- Hur HC, Lightfoot M, McMillin MG, Kho KA. Vaginal cuff dehiscence and evisceration: a review of the literature. Curr Opin Obstet Gynecol. 2016;28(4):297–303.
- Croak AJ, Gebhart JB, Klingele CJ, Schroeder G, Lee RA, Podratz KC. Characteristics of patients with vaginal rupture and evisceration. Obstet Gynecol. 2004;103(3):572–6.
- Chen MYM, Zagoria RJ, Dyer RB. Radiologic findings in acute urinary tract obstruction. J Emerg Med. 1997;15(3):339–43.
- Katabathina VS, Restrepo CS, Betancourt Cuellar SL, Riascos RF, Menias CO. Imaging of oncologic emergencies: what every radiologist should know. Radiographics. 2013;33(6):1533–53.
- Cervantes A, Chirivella I. Oncological emergencies. Ann Oncol. 2004;15(Suppl 4):iv299–306.
- 56. Chung KJ, Park BHB, Park BHB, Lee JH, Kim WJ, Baek M, et al. Efficacy and safety of a novel, double-layered, coated, selfexpandable metallic mesh stent (uventaTM) in malignant ureteral obstructions. J Endourol. 2013;27(7):930–5.
- 57. Misra S, Coker C, Richenberg J. Percutaneous nephrostomy for ureteric obstruction due to advanced pelvic malignancy: have we got the balance right? Int Urol Nephrol. 2013;45(3):627–32.
- 58. Caterino JM, Adler D, Durham DD, SCJ Y, Hudson MF, Bastani A, et al. Analysis of diagnoses, symptoms, medications, and admissions among patients with cancer presenting to emergency departments. JAMA Netw Open. 2019;2(3):e190979.
- Vandyk AD, Harrison MB, Macartney G, Ross-White A, Stacey D. Emergency department visits for symptoms experienced by oncology patients: a systematic review. Support Care Cancer. 2012;20(8):1589–99.
- Lefkowits C, Duska L. Opioid use in gynecologic oncology; balancing efficacy, accessibility and safety: an SGO clinical practice statement. Gynecol Oncol. 2017;144(2):232–4.
- Hacker KE, Reynolds RK, Uppal S. Ongoing strategies and updates on pain management in gynecologic oncology patients. Gynecol Oncol. 2018;149(2):410–9.
- Swarm R, Abernethy AP, Anghelescu DL, Benedetti C, Blinderman CD, Boston B, et al. NCCN Adult Cancer Pain. Adult cancer pain. J Natl Compr Cancer Netw. 2010;8(9):1046–86.
- Anekar AA, Cascella M. WHO Analgesic Ladder. 2020 May 17. In: StatPearls [Internet]. Treasure Island: StatPearls Publishing; 2020. PMID: 32119322.
- Page R, Blanchard E. Opioids and cancer pain: patients' needs and access challenges. J Oncol Pract. 2019;15(5):229–31.
- 65. Caraceni A, Hanks G, Kaasa S, Bennett MI, Brunelli C, Cherny N, European Palliative Care Research Collaborative (EPCRC); European Association for Palliative Care (EAPC), et al. Use of opi-

oid analgesics in the treatment of cancer pain: evidence-based recommendations from the EAPC. Lancet Oncol. 2012;13(2):e58-68.

- 66. Lutz S, Balboni T, Jones J, Lo S, Petit J, Rich SE, et al. Palliative radiation therapy for bone metastases: update of an ASTRO evidence-based guideline. Pract Radiat Oncol. 2017;7(1):4–12.
- 67. Ostby SA, Evans JG, Smith HJ, Boitano TKL, Toboni MD, Heimann MA, et al. Reducing emergency department (ED) computed tomography (CT) utilization in women treated for gynecologic cancers. Gynecol Oncol. 2020;156(2):288–92.
- Shaikh DH, Baiomi A, Mehershahi S, Abbas H, Gongati S, Nayudu SK. Paclitaxel-induced bowel perforation: a rare cause of acute abdomen. Case Rep Gastroenterol. 2020;14(3):687–94. https://doi.org/10.1159/000510131.
- 69. Liang B, Lange SS, Massad LS, Dick R, Mills KA, Hagemann AR, et al. Do gynecologic oncology patients with severely diminished renal function and urinary tract obstruction benefit from ureteral stenting or percutaneous nephrostomy? Gynecol Oncol Rep. 2019;28:136–40. https://doi.org/10.1016/j.gore.2019.04.007.

Robert C. Weinschenk and Patrick P. Lin

Case Study

A 57-year-old female with no known primary cancer presents to the emergency department (ED) with a 3-week history of progressive right groin pain. She denies any associated trauma. She previously ambulated independently without assistive devices, but now requires a cane. Her pain is most significant when ambulating, but she admits to pain at night and at rest. Over-the-counter pain medication has been ineffective. She denies any numbness, tingling, or weakness.

On physical exam, she has no gross neurovascular deficits, but is unable to perform straight leg raise and has marked pain with passive internal rotation of the right hip. Radiographs show a large lytic lesion in the femoral neck, which combined with her pain, results in a Mirels' score of 12 (Fig. 33.1). Laboratory workup is remarkable for an elevated TSH. CT of the chest, abdomen, and pelvis shows an enlarged thyroid, and bone scan shows no additional sites of bony disease.

The patient underwent core needle biopsy of her femoral neck lesion and was diagnosed with metastatic follicular thyroid carcinoma. She underwent total hip arthroplasty followed by postoperative radiation. She additionally underwent a complete thyroidectomy.

Introduction

Orthopedic emergencies are common in oncologic patients because of the prevalence of cancer and secondary effects of oncologic treatment. While the anatomic considerations are the same as those in the general population, the altered biology and physiology consequent to cancer can increase the complexity of treatment. Providers must consider tumor his-

R. C. Weinschenk \cdot P. P. Lin (\boxtimes)

© Springer Nature Switzerland AG 2021

tology, natural history of underlying disease, disease stage, and overall prognosis to achieve optimal, individualized care [1]. With that said, some of the more prevalent emergencies seen in the musculoskeletal oncologic have common features, such that an organized and systematic approach will set the physician up for success.

Trauma

Although the term "trauma" is usually associated with high-impact injuries, it also includes harm that occurs with minor force, which may pertain more to oncologic patients. In general, the core principles of trauma apply to oncologic patients, but there are several important distinctions for this population.

Pathologic Fractures

Pathologic fractures comprise a large portion of orthopedic oncologic emergencies, as they occur in 9–25% of patients with bone metastases [1]. The most common site is the proximal femur. In addition to pain and tenderness, patients may present with ecchymosis, deformity, edema, and joint effusion. The occurrence of a pathologic fracture can be the result of a long, complex process. Once disease is present in bone, it manipulates local biology, induces osteoclastic bone resorption, and ultimately causes loss of structural integrity.

Recognizing fractures as pathologic may not always be straightforward. This is particularly true for patients who present to the emergency department (ED) without a cancer diagnosis. Many patients are elderly with presumed osteoporosis. X-rays may be misleading because comminution around a fracture can hide a pre-existing bone lesion (Fig. 33.2). Failure to diagnose a fracture as pathologic can lead to grave consequences, including inappropriate surgery, delays in cancer treatment, and even loss of the limb.



Orthopedics

Department of Orthopedic Oncology, The University of Texas MD Anderson Cancer Center, Houston, TX, USA e-mail: plin@mdanderson.org



Fig. 33.1 (a, b) Preoperative and postoperative radiographs, respectively, of a 57-year-old female with metastatic thyroid carcinoma to the right femoral neck. Based on peritrochanteric location, lytic nature,

large size, and significant pain clinically, Mirels' score of this patient is 12, and thus, prophylactic surgery was recommended

The history is critical. Certain diseases, such as breast and renal cancer, can present years after the initial diagnosis with metastatic bone disease as the first manifestation of relapse. The practitioner should inquire about pre-existing pain. A deep-seated, constant ache prior to fracture should raise suspicion of an insidious process in the bone. Systemic signs such as weight loss, night sweats, and fatigue are relevant clues. The mechanism of injury can be revealing as minimal trauma, such as rolling over in bed, can result in pathologic fractures. In general, a break that occurs with a force equivalent to (or less than) a fall from a standing height suggests a pathologic fracture.

Most pathologic fractures are best treated surgically. The pain relief from surgical stabilization is far superior to the analgesia of opioids. If a patient has extensive metastatic disease, palliative surgery is still justified if the patient is expected to live for >1 month. Not offering an orthopedic consultation simply because a patient has advanced disease can be a mistake.

On the other extreme, it is also errant to perform surgery prematurely. Because metastases are common and pathologic fractures cause severe pain, it is easy to fall into the trap of rushing to surgery. Without clear cancer staging indicating metastatic disease, it can be grossly negligent to operate on a presumption of metastasis. Primary sarcomas can also be a cause of pathologic fractures. A common mistake is nailing a lesion, which later turns out to be primary osteosarcoma. This frequently results in amputation due to marrow and soft tissue contamination.

When the etiology of a pathologic fracture is uncertain, additional preoperative workup is necessary. Imaging includes whole body bone scan, chest radiographs, and CT scan of the chest, abdomen, and pelvis [2]. Not only will this help identify other sites of disease, but it will also help identify the most accessible lesion to biopsy [1]. The most common primary carcinomas to metastasize to bone are breast, prostate, thyroid, lung, and kidney [3]. To help make a specific diagnosis, laboratory workup may include serum and urine protein electrophoresis (SPEP/UPEP) to assess for myeloma, thyroid-stimulating hormone (TSH), and prostate-specific antigen (PSA). A complete blood count (CBC) and inflammatory markers (erythrocyte sedimentation rate (ESR) and C reactive protein (CRP)) may be helpful if osteomyelitis is in the differential diagnosis.



Fig. 33.2 A 43-year-old male with metastatic renal cell carcinoma, recent ground level fall with new inability to bear weight, with some preceding left hip pain. Initial X-rays show a displaced left femoral neck fracture, which could easily be mistaken for an osteoporotic frac-

ture (a). CT scan illustrates a discrete metastatic lesion in the femoral neck that predisposed to this fracture (b). Patient underwent a left hip hemiarthroplasty (c, d)

Oncologic patients undergoing preoperative workup require additional special considerations. Orthogonal radiographs should be obtained of the entire bone, including the joints above and below the lesion. This clarifies the region of interest, identifies concomitant areas of concern, and aids in preoperative planning. Some patients have anemia, thrombocytopenia, or neutropenia. Extensive osseous metastasis can also cause marked hypercalcemia requiring pharmacologic correction [4]. Coagulopathy may be present because of cachexia, poor nutrition, or liver disease. Bedridden patients may be prone to venous thromboembolism. Cross-matched units of packed red blood cells need to be available, since metastatic lesions can bleed profusely intraoperatively. A chest X-ray prior to surgery can help screen for malignant pleural effusions.

When immediate surgery is not possible or prudent, appropriate immobilization is necessary. In general, long

bone fractures should be immobilized in a well-padded splint that is not circumferential, to accommodate swelling. A joint above and joint below should be immobilized for mid-shaft fractures to prevent rotational stress. Exceptions to this include proximal humeral fractures and femoral fractures. Proximal humeral fractures can be immobilized in a cuff and collar or shoulder immobilizer. While some advocate Buck's traction for proximal femur fractures, current data do not indicate a clear benefit [5].

Surgical treatment of pathologic fractures has gradually become less invasive and stressful. In many instances, the fractures are amenable to intramedullary nailing, which can be placed percutaneously with minimal dissection. Nailing of proximal femoral metastases not involving the femoral head has been shown to be a reliable option with a 10% revision rate due to disease progression, breakage, and other reasons [6]. For pathologic fractures involving the femoral



Fig. 33.3 A 61-year-old male with metastatic renal cell carcinoma to the left femur, 3 months status post intramedullary nailing at an outside facility, with progressive disease, difficulty ambulating, and persistent

neck and head, hip arthroplasty may be more secure than nails (see Figs. 33.1 and 33.2). These operations can also be done now through relatively small (10–15 cm) incisions. In the uncommon scenario where extensive bone destruction is present, larger segmental endoprostheses may be required, as nails may be structurally inadequate and fail to relieve symptoms (see Fig. 33.3).

Impending Pathologic Fractures

Impending fractures represent a form of orthopedic emergency, and, when diagnosed properly, are a best-case scenario. Prophylactic surgery prior to fracture results in less stress to the patient and a shorter hospital stay [7]. Additional benefits include improved pain control, diminished opioid use, better quality of life, simpler surgery, decreased operating time, and less blood loss [1]. For these reasons, emergency providers must be able to recognize the signs and symptoms of impending fractures.

Pain is the most common sign indicating an impending pathologic fracture. The degree of pain often correlates with degree of structural compromise of the bone. Patients may at first attribute pain to a presumed injury, but notice later that pain does not improve with time. Pain can be exacerbated by weight-bearing but is classically present at rest and at night [8]. In pelvic and femoral lesions, back and lower extremity pain is often coupled with mechanical instability [4].

The most common way to gauge the risk of impending fracture is Mirels' score, which incorporates four categories: pain, location, size, and radiographic appearance (see

pain (**a**). The nail was removed and converted to a distal femur megaprosthesis (**b**), resulting in significant pain relief and improvement in function, as he was able to weight-bear immediately

Fig. 33.1). Each category is given 1–3 points, depending upon the severity: pain (mild, moderate, functional), location (upper extremity, lower extremity, peritrochanteric in the proximal femur), size relative to bone diameter (<1/3, 1/3-2/3, >2/3), and radiographic appearance (osteoblastic, mixed, lytic). The total score ranges from 4 to 12. With a Mirels' score of ≥9, the risk of pathologic fracture is >33% and prophylactic surgery is recommended. A Mirels' score of ≤7 typically is treated nonsurgically with radiation. A Mirels' score of 8 is left up to surgeon discretion [9].

Mirels' score is widely employed, in part because it uses simple radiographs. However, reliance upon twodimensional representation also limits its accuracy. In certain instances, CT scans can be helpful, since they provide three-dimensional detail regarding bone integrity and calcium density. While researchers have developed experimental methods of determining bone strength from CT scans that are more sensitive and specific in predicting pathologic fracture risk, these instruments are not yet available for widespread use [10].

In 22–30% of cases, patients with impending fractures do not have a known primary tumor at the time of presentation [2]. For adults over 40 years old, the differential diagnosis of a lytic lesion in bone includes metastatic disease, multiple myeloma, lymphoma, primary bone malignancy (sarcomas), destructive benign bone tumors (such as giant cell tumor of bone), and nonneoplastic conditions [8]. That being said, in this older age group, the most common malignant tumor of bone is metastatic disease, with the skeleton representing the third most frequent site of metastasis for carcinoma [2].

Open Fractures

While open injuries in cancer patients are mostly due to lowenergy mechanisms, they should not be dismissed lightly, and it is important to follow Advanced Trauma Life Support (ATLS) protocols. Of those with open fracture, 30% have more than one injury [11]. One should check airway, breathing, and circulation, followed by a thorough head to toe exam after removing constricting clothing. The deadly triad of acidosis, hypothermia, and coagulopathy is common in these patients, and it is imperative to correct these abnormalities quickly. Open injuries should be scrutinized for signs of contamination and exposed bone. If a limb is grossly deformed, it should be gently reduced prior to splinting. In the absence of an open injury, but with significant skin tenting, causing vascular compromise to the skin, a reduction maneuver is also worth performing [11]. Patients should be examined for signs of vascular injury, which may include decreased/ absent pulses, severe hemorrhage, expanding and pulsatile hematoma, bruit or thrill, numbness or neurologic deficit, decreased skin temperature, absent venous filling, absent pulse oximeter reading, or no capillary blanching [11]. If there is concern for vascular injury, a CT angiogram should be performed and a vascular surgery consultation obtained. All open wounds should be copiously irrigated prior to splinting, unless the patient is immediately being taken to the operating room. Due to environmental contamination, it is not recommended to obtain routine cultures from open wounds [11].

Early antibiotics are critical. Studies show that the most important factor in reducing infection rate for open fractures is early administration of intravenous antibiotics, ideally within 3 h of injury [12]. Open fractures are categorized by the Gustilo-Anderson classification: Type I (no gross contamination, laceration <1 cm, no significant muscle damage), Type II (+/- gross contamination, laceration 1-10 cm, no significant soft tissue crushing), and Type III (high energy, may have gross contamination, laceration >10 cm, extensive soft tissue injury). Type III is subdivided into types A (adequate soft tissue coverage), B (inadequate soft tissue coverage of bone), and C (vascular injury that requires repair). In general, Type I and II injuries are treated with a first-generation cephalosporin; for Type III injuries, add an aminoglycoside, and for farm injuries, add penicillin G. [13, 14]. All patients with open fractures must be asked about tetanus immunization history. Those who have had a booster in the past 5 years require no treatment. Those who have not had a booster in more than 5 years require tetanus toxoid plus human tetanus immunoglobulin (HTIG) if the wound is tetanus prone. Those who have not had a booster in more than 10 years or are immunocompromised require tetanus toxoid and HTIG [15].

Compartment Syndrome

Compartment syndrome is a problem that emergency providers must know how to identify, as it often demands immediate surgery. However, in the oncologic patient, compartment syndrome is usually not due to high-energy trauma but rather hematologic problems, and patients may not be good surgical candidates. The management thus has some unique features.

Compartment syndrome is a clinical diagnosis based on history and physical exam [16]. Fracture of the tibial diaphysis is the most common etiology, accounting for 36% of cases of acute compartment syndrome, with all four compartments frequently involved, but anterior compartment being most common [17]. Additional causes include blunt soft tissue injuries, burns, casts, venous thromboembolism, electromyelogram, extravasation of contrast, hematologic diseases, infection, hematomas, osteotomies, prolonged immobilization, and vascular procedures [18]. Pertinent laboratory values include creatine kinase, creatinine, BUN, potassium, urinalysis, and renal myoglobin. Rhabdomyolysis, which is frequent, causes renal insufficiency and hyperkalemia [18].

Classic findings include the 5Ps: pain, pulselessness, paralysis, paresthesia, and pallor. It should be stressed that excessive reliance upon the signs could lead to misdiagnosis since they may not be evident until late. Pain is usually the first symptom seen, typically due to ischemia and is often out of proportion to what is expected [19]. More specifically, pain occurs with passive stretch of the affected compartment (e.g., anterior compartment of tibia when ankle is passively plantar flexed). Marked firmness of the muscles, which is not one of the "Ps," is a reliable finding that is consistently present, but appreciation of muscle firmness requires some experience and can be difficult in obese or edematous limbs. Paralysis and pulselessness are usually very late findings after prolonged ischemia. Ironically, pulses may occasionally be bounding since the blood flow bypasses the capillaries in muscle, and this may lead the unwary clinician astray.

Although the diagnosis is usually evident from clinical examination, the findings are sometimes equivocal, especially for obtunded or mentally impaired patients. In these situations, an invasive diagnostic procedure can be performed to measure compartment pressures. A special manometer (i.e., Stryker Needle) is most commonly used. In the lower leg, all four compartments should be measured, ideally within 5 cm of the fracture site to improve diagnostic accuracy [20].

It is critical not to delay diagnosis or treatment. Duration of muscle ischemia predicts degree of muscle necrosis, and the damage to muscles becomes irreversible after 6 h of complete ischemia. On rare occasions, inflammatory response that ensues can become systemic and ultimately lead to multiple organ failure [19]. Once diagnosis is established, the patient is taken to the operating room emergently for fasciotomy. The fasciotomy wounds are typically left open until the edema and swelling subside.

While compartment syndrome usually occurs after trauma in the general population, cancer patients more likely have a different etiology. Many have thrombocytopenia or coagulopathy secondary to either the disease or its treatment. When the platelet count is less than 10,000 per microliter, patients are at risk for spontaneous bleeding [21], which could cause acute compartment swelling.

Oncologic patients may not be appropriate for fasciotomies. Neutropenia, defined as absolute neutrophil count (ANC) of $<1.5 \times 10^{9}/L$, is one relative contra-indication to surgery and predisposes patients to wound infections [22, 23]. However, some clinical judgment must be exercised, and fasciotomies may be performed with mild degrees of neutropenia. A more problematic issue is a delayed diagnosis of compartment syndrome. For patients without clear trauma and fracture, this may not be recognized until days after onset, particularly when patients are heavily sedated in the ICU. If compartment syndrome is diagnosed late, after myonecrosis has already occurred, it may be a grave mistake to perform fasciotomies and to leave wounds open, as the dead tissue becomes prone to severe, deep infections. In this situation, surgeons may opt to delay intervention and address muscle contractures at a later date.

Infection

Orthopedic infections represent a major category of oncologic emergencies because many patients are immunocompromised. Not only does this make patients more susceptible to contracting infections, but also, the severity of infections can be much greater. Infections can involve soft tissue, bones, joints, or all three. In some cases, infection is insidious and indolent, but in others, it can be sudden and rapid. A prevailing theme in infection is that acute phase reactants can be a clue not only to presence of infection, but also to timing. In general, ESR has a high specificity for infection and malignancy, rises within 24–48 h of onset, and gradually resolves. CRP begins rising 12–24 h after onset and peaks in 2–3 days. Procalcitonin begins rising within 3–4 h and peaks within 6–24 h [24].

Cellulitis

Cellulitis is a diffuse, permeative infection of soft tissues that does not produce a loculated fluid collection or drainage. It presents as erythema, warmth, pain, swelling, and sometimes fever. Cellulitis is common in the oncologic population for several reasons, including immunosuppression and frequency of minor procedures, such as venipuncture. The differential diagnosis includes septic arthritis, necrotizing fasciitis, lymphangitis, insect bite, foreign body reaction, abscess, osteomyelitis, inflammatory conditions, muscle hemorrhage/hematoma, Sweets syndrome, and venous thromboembolism [25–27]. Noninfectious inflammatory conditions can be especially difficult to distinguish from cellulitis. These include gout, pseudogout, tenosynovitis, degenerative joint diseases, and rheumatologic conditions [25], all of which may be exacerbated acutely by chemotherapy.

Cellulitis is often a diagnosis of exclusion. Examination is useful for determining the likelihood of joint involvement and the presence of fluctuance. X-rays may help to rule out osteomyelitis [27], while ultrasound, MRI, and/or CT may be necessary to rule out loculated fluid collections and involvement of bone or joint. MRI characteristically shows diffuse edematous changes, bright on T2-weighted images, in cellulitis. Patients should be treated with empiric antibiotics since blood cultures are positive in <5% of patients [27]. The offending organism is most often streptococci or Staphylococcus aureus, and initial antibiotics should target Gram-positive bacteria [27]. Drawing a line around the region of erythema with indelible ink is helpful to track response to therapy. Depending on the degree of neutropenia, hospital admission for broad-spectrum intravenous antibiotics may be indicated.

Necrotizing Fasciitis

Necrotizing fasciitis is an acute, life-threatening infection that even among healthy individuals has a high mortality rate. It is imperative that this disease process be recognized early to maximize chances of survival. It is characterized by rapidly spreading inflammation followed by necrosis of fascial planes and surrounding soft tissue [28]. As many as 60% of patients have underlying diabetes [28]. Other risk factors include alcoholism, end stage renal disease, cardiopulmonary disease, peripheral vascular disease, advanced age, cancer/chemotherapy, immunosuppression, malnutrition, trauma, surgery, IV drug abuse, and smoking [28, 29]. Comorbid conditions that correlate with mortality include cancer, renal disease, and congestive heart failure [30]. Metastatic carcinoma has also been reported to cause a soft tissue abscess that ultimately led to development of necrotizing fasciitis [31].

Recognizing early signs and symptoms of necrotizing fasciitis is crucial but can be difficult, as pain frequently precedes overt findings [28]. Systemic symptoms include fever, chills, hypotension, and tachycardia. While patients may not have frank altered mental status, family mem-

bers often report that the patient is not behaving normally. Common skin findings, in addition to erythema and induration, are blisters and bullae that at first drain serosanguinous fluid and later become hemorrhagic. If there is associated soft tissue gas, crepitus will be present. A characteristic frequent finding of this disease is "dishwater pus," which describes the thin, watery, foul-smelling fluid produced consequent to necrosis of superficial fascia and fat [32]. The most common site of infection is the lower extremity, followed by the upper extremity and trunk [33].

Laboratory tests include CBC, comprehensive metabolic panel (CMP), ESR, and CRP. A useful diagnostic tool is the Laboratory Risk Indicator for Necrotizing Fasciitis (LRINEC) Score, which has a reported PPV of 92% and NPV of 96% [34]. The LRINEC score considers CRP $(\geq 150 \text{ mg/L} - 4 \text{ points})$; white blood cell (WBC) count $(<15/mm^{3} - 0 \text{ points}, 15-25/mm^{3} - 1 \text{ point}, >25/mm^{3} - 2$ points); hemoglobin (>13 g/dL - 0 points, 11-13.5 g/dL -1 point, <11 g/dL - 2 points); sodium (<135 mmol/L - 1point); creatinine (>141/ μ mol/L - 2 points); and glucose (>10 mmol/L – 1 point). A total score of ≥ 6 is considered suspicious for the disease. While this may be true for the general population, the instrument must be used with caution and judiciously applied in the cancer population, which is replete with co-morbidities including chronic anemia, neutropenia, and renal insufficiency.

Plain films and CT scan classically show signs of gas in the soft tissue; however, gas tracking along fascial planes is only seen in 55% of CT scans. CT, however, will show asymmetric fascial thickening and fat stranding in 80% of cases [35]. A helpful bedside maneuver is the "finger test," whereby a 2-cm incision is made down to deep fascia and a gloved finger inserted to the base. A positive test is defined by dishwater pus, absence of skin bleeding, and lack of tissue resistance to blunt dissection [28]. The gold standard for diagnosis, however, remains fascial biopsy in the operating room.

Emergent surgery is required for any meaningful chance of survival. Prior to surgery, broad-spectrum antibiotics may be initiated to include coverage of streptococci, anaerobes, Gram-negative organisms, and Gram-positive bacilli [29]. More than 80% of cases are Type 1, defined as polymicrobial, whereas the remainder are Type 2 (Group A beta-hemolytic *Streptococcus* sp.) and Type 3 (marine *Vibrio* sp.) [28].

Other Soft Tissue Infections (Abscess, Bursitis, Tenosynovitis)

Unlike cellulitis, a deep tissue abscess is characterized by the presence of a purulent fluid-filled cavity, which can be documented by ultrasound, MRI, or CT scan. Drainage of the fluid, whether by aspiration, percutaneous drain placement, or surgery, is an essential aspect of treatment, since these infections typically do not resolve with antibiotics alone. It is important that fluid samples be obtained prior to initiation of antibiotics so that the organism can be isolated and tested for antibiotic susceptibility.

Synovial-lined spaces around tendons and bursa can potentially become infected and produce abscesses that need drainage or surgical debridement. Certain areas are especially prone to producing infected bursitis, including the olecranon and prepatellar bursas [36]. Although uncommon, flexor tenosynovitis in the hand is a true emergency that needs immediate attention, since this can result in rapid scarring and loss of function of the digit. The diagnosis is often made clinically based on examination and Kanavel's cardinal signs – flexion of the finger, fusiform swelling, tenderness along the flexor sheath, and pain with passive extension. Treatment is prompt surgical drainage, irrigation, and antibiotics [37].

Septic Arthritis

Septic arthritis can be defined as joint infection caused by pathogenic inoculation of the joint either directly or by hematogenous spread [38]. Risk factors for septic arthritis include presence of a prosthesis, intravenous drug use, alcoholism, diabetes, prior corticosteroid injection, and cutaneous ulcers. Immunosuppression resulting from chemotherapy or organ transplant can make patients particularly prone to septic arthritis [1].

A common proverb that underscores the urgency of septic arthritis is, "Time is cartilage." With time, bacterial toxins promote degradation of articular cartilage, mediated by proteolytic enzymes [1]. It is therefore imperative that it be diagnosed and treated in a timely manner, as irreversible joint damage and septicemia can even be seen in patients managed appropriately [38]. On physical exam, patients often are febrile with warmth and erythema about the joint with associated guarding. If the lower extremity is involved, they may have an antalgic gait. Patients tend to assume a position that minimizes pressure within the joint, which in the hip is a flexed, abducted, and externally rotated position, and in the knee or elbow is a slightly flexed position. It should be noted that in immunocompromised patients, fever and guarding may be diminished or absent [1].

Appropriate laboratory workup is critical. Initial labs should include CBC, CMP, ESR, and CRP. The gold standard for diagnosis of a septic joint is aspiration. This should be done after radiographs to evaluate for fracture or other pathologic processes. While the classic teaching is that synovial fluid with white cell count of >50,000 /mm³ is synonymous with septic arthritis, this is an oversimplification. Coutlakis, et al. showed that a count of <50,000 was 5% sensitive, 50,000–100,000 was 47% sensitive, and >100,000 was 77% sensitive. In this study, crystalline arthropathies and rheumatoid arthritis accounted for 81% of patients with white cell count in the 15,000–50,000 range, and 5/7 cases of septic arthritis with white cell count <20,000 grew atypical organisms [39]. Furthermore, patients with a polymorphonuclear (PMN) cell count of \geq 90% are far more likely to have a septic joint than those <90% [40].

Antibiotics should be held until after joint aspiration in all medically stable patients. Following aspiration, initial antibiotics target common pathogens, with *S. aureus* and *Streptococcus* species accounting for more than 90% of cases [38]. That being said, atypical species such as Gram-negative organisms, *Mycobacterium*, and fungi can be seen in immunocompromised patients [1]. Antibiotics can be adjusted for susceptibilities and are typically continued for approximately 4–6 weeks [1]. Standard surgical care includes open or arthroscopic irrigation and debridement. In oncologic patients who are neutropenic or immunocompromised, alternatives include serial joint aspiration and bedside lavage.

Periprosthetic joint infections (PJIs) represent a different entity than septic arthritis of the native joint. However, timing remains critical. In acute infections, <4 weeks postoperatively, it may be possible to perform a surgical irrigation and debridement with modular component exchange (i.e., replacing easily removable parts such as plastic liners but leaving fixed hardware in place). In chronic infections, surgery entails removal of all components, insertion of an antibiotic spacer, an extended course of IV antibiotics, and a revision prosthesis.

On physical exam, PJIs can have similar findings to native septic joints. Additionally, there may be wound-related complications, such as dehiscence or drainage. However, PJIs sometimes show remarkably few signs, especially for chronic infections. While acute infections commonly show signs of rapid onset swelling, warmth and tenderness, chronic infections may be more subtle, with gradual, insidious pain, and no erythema or fever [41]. Radiographs are usually normal, but sometimes will show loosening of a previously well-fixed implant (especially if <5 years postoperative), subperiosteal elevation, or transcortical sinus tract [42]. Common risk factors include recent bacteremia, multiple prior surgeries, prior PJI, smoking, diabetes mellitus, cancer, inflammatory arthropathy, and immunosuppressive drugs [41, 42].

Laboratory cutoff values for PJI differ with respect to time. In patients who are <6 weeks postoperative, cutoffs are as follows: ESR has none, CRP >100 mg/L, synovial WBC count >10,000 cells/ μ L, PMN >90%. For patients >6 weeks postoperative, these values differ: ESR >30 mm/h, CRP >10 mg/L, synovial WBC count >3000 cells/ μ L, PMN >80% [42]. Diagnostic criteria are broken into major and minor. Major criteria include two positive cultures of the same organism or a sinus tract that communicates with the joint. Minor criteria include elevated CRP or D-dimer, elevated ESR, elevated synovial WBC, positive synovial alpha-defensin, elevated synovial WBC, positive synovial CRP. To confidently diagnose PJI, either 1 major or 6 minor criteria must be present [43]. Sinus tracts and open wounds should not be cultured in the ED, as they are more likely to grow colonized, nonpathogenic bacteria that confuse treatment goals. Following aspiration, empiric antibiotics should include MRSA coverage. In the uncommon scenario where the patient is unstable and septic, broad-spectrum antibiotics should be initiated prior to joint aspiration and blood cultures should be drawn [41].

Osteomyelitis

Osteomyelitis (OM) is defined as inflammation of the bone caused by an infecting organism [44]. Acute hematogenous OM is the most prevalent form. Patients typically present with fever and malaise. While WBC may be normal, inflammatory markers are usually elevated. Blood cultures should always be drawn, as the causative organism can be identified in approximately 50% of patients [44]. S. aureus accounts for the highest percentage of acute cases [45]. Radiographs often are unremarkable but can show soft tissue swelling after 1-3 days and periosteal elevation or bony destruction after 10-12 days [44]. MRI can be useful for diagnosis of OM and identification of associated abscesses. Empiric antibiotic coverage should include Gram-positive bacteria. Gram-negative bacteria, and potentially resistant organisms, depending on the patient's history [45]. The triad of fever, elevated CRP, and elevated absolute neutrophil count (ANC) correctly predicts methicillin-resistant S. aureus (MRSA) over methicillinsensitive S. aureus (MSSA) in 87% of cases [46]. In leukemic patients, atypical bacteria or fungi have been isolated [47, 48]. Definitive treatment is with antibiotics. Some patients, particularly those with abscesses, may also require surgery.

Chronic OM is more indolent. Patients may have intermittent exacerbations over a course of years. Like acute cases, inflammatory markers are more sensitive than WBC count, which is elevated in only 35% of cases [44]. Those with chronic OM can have open wounds or sinus tracts communicating with deeper infection. It is not recommended to culture such wounds, as the isolates do not typically correlate with organisms that grow from an intraoperative bone biopsy [44]. In addition to X-rays, CT scan may help identify a sequestrum (infected, dead bone surrounded by new bone growth). Management requires a multidisciplinary approach, as the infection is typically difficult to eradicate, and associated comorbidities may preclude an aggressive debridement. Chronic OM can undergo malignant degeneration, most often as a squamous cell carcinoma (i.e., Marjolin ulcer) [49]. The latent period averages 27–30 years, with acute cases being very rare. Marjolin ulcers are characterized by drainage, foul odor, and a large, exophytic mass. Imaging should include X-rays, MRI, and a CT chest to evaluate for metastases. Diagnosis is confirmed by biopsy, and definitive treatment is wide excision or amputation [49].

Arthroplasty

Joint arthroplasty as a reconstructive modality has grown steadily with time. In years past, malignant primary bone tumors were routinely treated with amputation, but since the advent of effective chemotherapy, most sarcomas are now amenable to limb salvage [50]. The success of endoprosthetic reconstruction for primary bone sarcomas has led to its expanded use in benign and metastatic bone disease. While endoprostheses have consistently shown acceptable outcome in oncologic patients, the reported complication rates are 5–10 times higher than for routine, nononcologic joint arthroplasty [50].

Periprosthetic Fractures

Periprosthetic fractures are common but not always obvious since fractures may be minimally displaced. They are important to recognize early. The most significant negative prognostic indicator relating to early outcomes in periprosthetic hip fractures is delay to surgery [51]. The risk of periprosthetic fracture increases with chronic infection, poor bone quality, stress shielding, osteolysis, and oncologic resections [50, 52]. For oncologic patients, progression or relapse of disease in the bone surrounding the stem must be considered.

Common findings are pain, gross deformity, and swelling. A detailed exam of the entire extremity is important to rule out concomitant bony or neurovascular injury. Full-length orthogonal radiographs should always be obtained. Despite metallic streak artifacts, CT scans can be helpful when radiographs are equivocal. The most common classification of periprosthetic hip fractures is the Vancouver classification: A (trochanteric), B (along the femoral stem, with 3 subtypes), and C (diaphyseal fracture distal to stem) (Fig. 33.4) [53]. Periprosthetic distal femur fractures are most commonly classified according to the Lewis and Rorabeck classification: I (nondisplaced fracture, intact prosthesis), II (displaced fracture, intact prosthesis), and III (loose or failing prosthesis) [54]. While these classification systems are designed for primary joints, the concepts of fracture location relative to the implant can be used for oncologic reconstructions as well [55].

Prosthetic Dislocation

The most common area for prosthetic dislocation is in the hip. In fact, dislocation accounts for >20% of revision hip surgery [56]. Dislocations can be attributed to one of three factors. Patient factors include older age, neurologic disease, cognitive impairment, spinal disease, and weak abductors [56, 57]. Implant factors include small head size and decreased offset. Surgeon factors include surgical approach (posterior approach most likely to dislocate) and component positioning.

Important details of the history include surgical approach, mechanism of dislocation, number of previous dislocations, and patient compliance with postoperative restrictions [58]. Location of surgical scars can serve as clues as to which approach was used. Posterior dislocations frequently occur when patients stand from a seated position. They result in an adducted, flexed, internally rotated, and shortened limb. Anterior dislocations result in abduction and external rota-



Fig. 33.4 (a, b) Preoperative and postoperative radiographs, respectively, of an 86-year-old female with multiple myeloma who sustained a pathologic periprosthetic femur fracture of the Vancouver C variety, for which she underwent open reduction with internal fixation

tion with either extension (pubic type) or flexion (obturator type) [58]. Evaluation of neurovascular status is important, as up to 10% of posterior dislocations can result in sciatic nerve palsy. Since infection can predispose to dislocation, laboratory workup should include a CBC, ESR, and CRP.

If an experienced emergency provider has adequate resources, a closed reduction may be attempted. While there are many techniques, the authors prefer the Waddell technique for posterior dislocations, whereby an assistant stabilizes the pelvis, and the provider straddles the patient's leg with the patient's hip and knee flexed, and exerts traction, with a combination of adduction and internal/external rotation [59]. Conscious sedation in the ED to reduce dislocated hips has been proven as a safe, effective method [60]. However, modular prostheses of the proximal femur in tumor patients have shown a re-dislocation rate following closed reduction of 58% compared to 11% of those that underwent an open reduction [61].

Complex arthroplasty for tumor patients is often performed with one of two methods to confer better stability: constrained liners or dual mobility systems. Constrained liners contain a metal ring to help stabilize the articulation between the head and the cup (Fig. 33.5). Dual mobility cups consist of a small metal/ceramic femoral head within a larger polyethylene head, which articulates with the acetabular component [58]. These are prone to what is called an intraprosthetic dislocation, whereby the polyethylene dissociates from the metal/ceramic head. While there are reports of successful closed reduction of dislocated constrained liners, it is typically recommended that the orthopedic surgeon perform open reduction of dislocations involving constrained liners or dual mobility cups [58, 62].

Pelvis

Patients with tumors of the bony pelvis and/or surrounding soft tissue represent a unique entity. The pelvis is the common pathway for all neurovascular structures supplying the lower extremities, genitalia, and perineum, thereby making surgery in this area complex and prone to complications. Hemipelvectomies performed for pelvic sarcomas are categorized by the area that is excised, but more broadly, are grouped as either internal (limb is preserved) or external (limb is amputated) [63]. One study showed a complication rate of 53% for external hemipelvectomies, most commonly due to wound infection and flap necrosis (Fig. 33.6) [64]. Other complications include ureteral injury, bladder fistula, strangulated incisional hernia, colocutaneous fistula, small bowel fistula, wound dehiscence, myocardial infarction, recurrent pulmonary emboli, and upper gastrointestinal bleed. In another study looking at both internal and external hemipelvectomies, wound infection occurred in 61.7% of cases [65]. Since these surgeries involve manipulation and/or reconstruction of large vessels, a possible dysvascular limb postoperatively should not be overlooked.



Fig. 33.5 A 53-year-old male with history of right pelvic chondrosarcoma 11 years status post internal hemipelvectomy and pelvic reconstruction with allograft prosthetic reconstruction with constrained liner

total hip arthroplasty. Following a fall, the patient experienced a dislocation (a) that was revised to a new constrained liner in the operating room (b)



Fig. 33.6 A 15-year-old male who developed a radiation-induced pelvic osteosarcoma following treatment for neuroblastoma. He underwent a right external hemipelvectomy with reconstruction of the pelvic ring using the tibia of his amputated limb (a). Postoperatively, he developed a wound dehiscence and infection that required several months of antibiotics and wound care (b)

References

- Lewis VO, Luna JT. Orthopedic emergencies in oncology. In: Robotin M, Olver I, Girgis A, editors. When cancer crosses disciplines: a physician's handbook. 1st ed. London: Imperial College Press; 2009. p. 569–601.
- Piccioli A, Maccauro G, Spinelli MS, Biagini R, Rossi B. Bone metastases of unknown origin: epidemiology and principles of management. J Orthop Traumatol. 2016;16(2):81–6.
- Galasko C. The anatomy and pathways of skeletal metastases. In: Weiss L, Gilbert A, editors. Bone metastases. Boston: GK Hal; 1981. p. 49–63.
- Coleman RE. Clinical features of metastatic bone disease and risk of skeletal morbidity. Clin Cancer Res. 2006;12(Suppl 20):6243–9.
- Handoll HH, Queally JM, Parker MJ. Pre-operative traction for hip fractures in adults. Cochrane Database Sys Rev. 2011;7(12):CD000168.
- Chafey DH, Lewis VO, Satcher RL, Moon BS, Lin PP. Is a cephalomedullary nail durable treatment for patients with metastatic peritrochanteric disease? Clin Orthop Relat Res. 2018;476(12):2392–401.
- Beals RK, Lawton GD, Shell WE. Prophylactic internal fixation of the femur in metastatic breast cancer. Cancer. 1971;28(5):1350–4.
- Weber KL. Evaluation of the adult patient (aged >40 years) with a destructive bone lesion. J Am Acad Orthop Surg. 2010;18(3):169–79.
- Mirels H. Metastatic disease in long bones: a proposed scoring system for diagnosing impending pathological fractures. Clin Orthop Relat Res. 1989;249:256–64.
- Nazarian A, Entezari V, Zurakowski D, Calderon N, Hipp JA, Villa-Camacho JC, et al. Treatment planning and fracture prediction in patients with skeletal metastasis with CT-based rigidity analysis. Clin Cancer Res. 2015;21(11):2514–9.

- Rajasekran S, Devendra A, Perumal R, Dheenadhayalan J, Sundrarajan SR. Initial management of open fractures. In: Court-Brown CM, Heckman JD, McQueen MM, Ricci WM, Tornetta P, McKee MD, editors. Rockwood and Green's fractures in adults. 8th ed. Philadelphia: Wolters Kluwer Health; 2015. p. 353–96.
- Patzakis MJ, Wilkins J. Factors influence infection rate in open fracture wounds. Clin Orthop Relat Res. 1989;243:36–40.
- Gustilo RB. Anderson JT. Prevention of infection in the treatment of one thousand and twenty-five open fractures of long bones: retrospective and prospective analyses. J Bone Joint Surg Am. 1976;58:453–8.
- Egol KA, Koval KJ, Zuckerman JD. Handbook of fractures. 4th ed. Philadelphia: Wolters Kluwer Health; 2010.
- Bleck TP. *Clostridum tetani* (tetanus). In: Mandell G, Douglas R, Bennett J, editors. Principles and practice of infectious diseases. 6th ed. Philadelphia: Churchill Livingstone; 2005.
- Maher M, Mauffrey C. Diagnostic dilemma for the orthopedic surgeon. In: Mauffrey C, Hak DJ, Martin III MP, editors. Compartment syndrome: a guide to diagnosis and management. Cham: Springer Open; 2020. p. 1–8. https://doi.org/10.1007/978-3-030-22331-1. Accessed 21 Sep 2020.
- McQueen MM, Gaston P, Court-Brown CM. Acute compartment syndrome. Who is at risk? J Bone Joint Surg Br. 2000;82(2):200–3.
- Long B, Koyfman A, Gottlieb M. Evaluation and management of acute compartment syndrome in the emergency department. J Emerg Med. 2019;56(4):386–97.
- McQueen MM. Acute compartment syndrome. In: Court-Brown CM, Heckman JD, McQueen MM, Ricci WM, Tornetta P, McKee MD, editors. Rockwood and Green's fractures in adults. 8th ed. Philadelphia: Wolters Kluwer Health; 2015. p. 895–913.
- Heckman MM, Whitesides TE, Grewe SR, Rooks MD. Compartment pressure in association with closed tibial fractures. The relationship between tissue pressure, compartment, and the distance from the site of the fracture. J Bone Joint Surg Am. 1994;76(9):1285–92.

- Nagrebetsky A, Al-Samkari H, Davis NM, Kuter DJ, Wiener-Kronish JP. Perioperative thrombocytopenia: evidence, evaluation, and emerging therapies. Br J Anaesth. 2019;122(1):19–31.
- Viscoli C, Varnier O, Machetti M. Infections in patients with febrile neutropenia: epidemiology, microbiology, and risk stratification. Clin Infect Dis. 2005;40(Suppl 4):240–5.
- Newburger PE, Dale DC. Evaluation and management of patients with isolated neutropenia. Semin Hematol. 2013;50(3):198–206.
- Markanday A. Acute phase reactants in infections: evidencebased review and a guide for clinicians. Open Forum Infect Dis. 2015;2(3):ofv098.
- Patel DP, Emmanual NB, Stevanovic MV, Matcuk GR Jr, Gottsegen CJ, Forrester DM, White EA. Hand infections: anatomic, types and spread of infection, imaging findings, and treatment options. Radiographics. 2014;34(7):1968–86.
- Hayeri MR, Ziai P, Shehata ML, Teytelboym OM, Huang BK. Softtissue infections and their imaging mimics: from cellulitis to necrotizing fasciitis. Radiographics. 2016;36(6):1888–910.
- Zinn S, O'Donnell JA. Cellulitis and skin infections associated with bites. In: Skolnik NS, Albert RH, editors. Essential infectious disease topics for primary care. Totowa NJ: Humana Press; 2008. p. 189–202.
- Bellapianta JM, Ljungquist K, Tobin E, Uhl R. Necrotizing fasciitis. J Am Acad Orthop Surg. 2009;17(3):174–82.
- Taviloglu K, Yanar H. Necrotizing fasciitis: strategies for diagnosis and management. World J Emerg Surg. 2007;2:19.
- Childers BJ, Potyondy LD, Nachreiner R, Rogers FR, Childers ER, Oberg KC, et al. Necrotizing fasciitis: a fourteen-year retrospective study of 163 consecutive patients. Am Surg. 2002;68(2):109–16.
- Hendrickson S, Bystrzonowski N, Kokkinos C, Butler P. Necrotising fasciitis caused by metastatic endometrial cancer: a rare cause of a life-threatening condition. Ann R Coll Surg Engl. 2017;99(2):e72–4.
- Wilkerson R, Paull W, Coville FV. Necrotizing fasciitis. Review of the literature and case report. Clin Orthop Relat Res. 1987;216:187–92.
- 33. Kaul R, McGreer A, Low DE, Green K, Schwartz B. Populationbased surveillance for group a streptococcal necrotizing fasciitis: clinical features, prognostic indicators, and microbiologic analysis of seventy-seven cases. Ontario Group A Streptococcal Study. Am J Med. 1997;103(1):18–24.
- 34. Wong CH, Khin LW, Heng KS, Tan KC, Low CO. The LRINEC (Laboratory Risk Indicator for Necrotizing Fascitis) score: a tool for distinguishing necrotizing fasciitis from other soft tissue infections. Crit Care Med. 2004;32(7):1535–41.
- Wysoki MG, Santora TA, Shah RM, Friedman AC. Necrotizing fasciitis: CT characteristics. Radiology. 1997;203(3):859–63.
- Hanrahan JA. Recent developments in septic bursitis. Curr Infect Dis Rep. 2013;15(5):421–5.
- Giladi AM, Malay S, Chung KC. A systemic review of the management of acyte pyogenic flexor tenosynovitis. J Hand Surg Eur Vol. 2015;40(7):720–8.
- Mathews CJ, Coakley G. Septic arthritis: current diagnostic and therapeutic algorithm. Curr Opin Rheumatol. 2008;20(4):457–62.
- Coutlakis PJ, Roberts WN, Wise CM. Another look at synovial fluid leukocytosis and infection. J Clin Rheumatol. 2002;8(2):67–71.
- Margaretten ME, Kohlwes J, Moore D, Bent S. Does this adult patient have septic arthritis? JAMA. 2007;297(13):1478–88.
- 41. Luthringer TA, Fillingham YA, Okroj K, Ward EJ, Della Valle C. Periprosthetic joint infection after hip and knee arthroplasty: a review for emergency care providers. Ann Emerg Med. 2016;68(3):324–34.

- Zmistowski B, Della Valle C, Bauer TW, Malizos KN, Alavi A, Bedair H, et al. Diagnosis of periprosthetic joint infection. J Orthop Res. 2014;32(Suppl 1):S98–107.
- 43. Parvizi J, Tan TL, Goswani K, Higuera C, Della Valle C, Chen AF, Shohat N. The 2018 definition of periprosthetic hip and knee infection: an evidence-based and validated criteria. J Arthroplast. 2018;33(5):1309–14.
- 44. Dabov GD. Osteomyelitis. In: Canale ST, Azar FM, Beaty JH, Campbell WC, editors. Campbell's operative orthopaedics. 13th ed. Philadelphia: Elsevier; 2017. p. 764–87.
- Kolinsky DC, Liang SY. Musculoskeletal infections in the emergency department. Emerg Med Clin North Am. 2018;36(4): 751–66.
- 46. Dietrich LN, Reid D, Doo D, Fineberg NS, Doo D, Fineberg NS, et al. Predicting MSSA in acute hematogenous osteomyelitis in a setting with MRSA prevalence. J Pediatr Orthop. 2015;35(4):426–30.
- 47. Doganis D, Baka M, Tsolia M, Pourtsidis A, Lebessi E, Varvoutsi M, et al. Multifocal Aeromonas Osteomyelitis in a Child with Leukemia. Case Rep Infect Dis. 2016;8159048:2016.
- Ma L, Tong H, Ruan L, Ling Z, Ren Y, Zhou X. Successful treatment of Candida tropicalis osteomyelitis with Micafungin in a leukemia patient. IDCases. 2016;6:109–11.
- Panteli M, Puttaswamaiah R, Lowenberg DW, Giannoudis PV. Malignant transformation in chronic osteomyelitis: recognition and principles or management. J Am Acad Orthop Surg. 2014;22(9):586–94.
- Savvidou OD, Sakellariou VI, Megaloikonomos PD, Mavrogenis AF, Papgelopoulos PJ. Periprosthetic fractures in megaprostheses: algorithmic approach to treatment. Orthopedics. 2017;40(3):e387–94.
- Griffiths EJ, Cash DJ, Kalra S, Hopgood PJ. Time to surgery and 30-day morbidity and mortality of periprosthetic hip fractures. Injury. 2013;44(12):1949–52.
- Marshall RA, Weaver MJ, Sodickson A, Khurana B. Periprosthetic femoral fractures in the emergency department: what the orthopedic surgeon wants to know. Radiographics. 2017;37(4):1202–17.
- Duncan CP, Masri BA. Fractures of the femur after hip replacement. Instr Course Lect. 1995;44:293–304.
- Rorabeck CH, Taylor JW. Classification of periprosthetic fractures complicating total knee arthroplasty. Orthop Clin North Am. 1999;30(2):209–14.
- 55. Gebhart M, Shmelinsky F. Management of periprosthetic fractures in patients treated with a megaprosthesis for malignant bone tumours around the knee. Acta Orthop Belg. 2012;78(4): 558–63.
- Brooks PJ. Dislocation following total hip replacement: causes and cures. Bone Joint J. 2013;95-B(Suppl 11A):67–9.
- 57. Fessy MH, Putman S, Viste A, Isida R, Ramdane N, Ferreira A, Leglise A, SFHG, et al. What are the risk factors for dislocation in primary total hip arthroplasty? A multicenter case-control study of 128 unstable and 438 stable hips. Orthop Traumatol Surg Res. 2017;103(5):663–8. Erratum in: Orthop Traumatol Surg Res. 2017;103(7):1137.
- Dawson-Amoah K, Raszewski J, Duplantier N, Waddell BS. Dislocation of the hip: a review of types, causes, and treatment. Ochsner J. 2018;18(3):242–52.
- Waddell BS, Mohamed S, Glomset JT, Meyer MS. A detailed review of hip reduction maneuvers: a focus on physician safety and introduction of the waddell technique. Orthop Rev. 2016;8(1):10–5.
- Frymann SJ, Cumberbatch GL, Stearman AS. Reduction of dislocated hip prosthesis in the emergency department using conscious sedation: a prospective study. Emerg Med J. 2005;22(11):807–9.

- 61. Puchner SE, Funovics PT, Hipfl C, Dominkus M, Windhager R, Hofstaetter JG. Incidence and management of hip dislocation in tumour patients with a modular prosthesis of the proximal femur. Int Orthop. 2014;38(8):1677–84.
- 62. Birdwell S, Wikle E. Closed reduction of constrained total hip arthroplasty in the ED. J Emerg Med. 2011;40(2):162–6.
- Enneking WF, Dunham WK. Resection and reconstruction for primary neoplasms involving the innominate bone. J Bone Joint Surg Am. 1978;60:731–46.
- Apfelstaedt JP, Driscoll DL, Spellman JE, Velez AF, Gibbs JF, Karakousis CP. Complications and outcome of external hemipelvectomy in the management of pelvic tumors. Ann Surg Oncol. 1996;3(3):304–9.
- 65. Guder WK, Hardes J, Gosheger G, Henrichs MP, Nottrott M, Streitburger A. Analysis of surgical and oncological outcome in internal and external hemipelvectomy in 34 patients above the age of 65 years at a mean follow-up of 56 months. BMC Musculoskelet Disord. 2015;16:33.

pdates

Joanna Harp and Alina Markova

Case Study

A 70-year-old man with metastatic melanoma started on ipilimumab with nivolumab 11 days ago presents to the emergency department with a 6-day history of a painful rash and mucositis. He reports initial sores in the oral and ocular mucosa and subsequent development of pain with urination. He then developed a red painful rash that continued to spread over the next few days. He denies any other new medications and known allergies include sulfa medications.

On exam he is well-appearing with stable vital signs with widespread erythematous macules and papules on 50% of his body some with central duskiness that resemble targetoid lesions on <10% of his body. He has hemorrhagic crusts and ulceration of the lips and oropharynx, as well as conjunctival injection and periurethral erythema. With lateral pressure on the dusky areas of the rash, he develops shearing of skin.

Metabolic panel reveals normal electrolytes, serum bicarbonate of 23 mEq/L, glucose of 113 mg/dL, BUN of 30 mg/ dL (H), creatinine of 1.1 mg/dL, albumin of 3.4 g/dL (L), AST of 172 U/L (H), and ALT of 199 U/L (H). Mycoplasma respiratory PCR and serum IgM and IgG are ordered. A punch biopsy is performed, revealing a lichenoid and subepidermal vesicular dermatitis with full-thickness epidermal necrosis.

The patient is diagnosed with Stevens-Johnson syndromelike rash, CTCAE grade 3, due to ipilimumab/nivolumab. He is admitted to the intensive care unit and started on cyclosporine 2.5 mg/kg twice daily for 10 days and then tapered by 1 mg/kg/week, as well as topical cyclosporine drops to both eyes, dexamethasone swish and spit for his mouth, topi-

J. Harp (🖂)

A. Markova

Center, New York, NY, USA

Department of Dermatology, Memorial Sloan Kettering Cancer

cal steroids to his glans, and Foley catheter. Denuded areas are covered with non-adhesive silver nylon dressings.

During his hospitalization, he does not have further fevers, but develops additional areas of denudation (<10% BSA). After day 5 of cyclosporine, he no longer develops new areas of rash, by day 15 his rash begins to hyperpigment, by day 22 his mucositis resolves, transaminases downtrend, and he is discharged.

Introduction

Patients with cancer can present with a wide range of common, serious, and at times life-threatening dermatologic conditions related to their underlying malignancy or to its treatment. Additionally, they may be at increased risk for infections, including cutaneous infections, secondary to immunosuppression. Recognition of the morphology of skin lesions (i.e., color, texture, shape, distribution, etc.) is an important step in the evaluation of these patients as it can greatly aid in formulating a differential diagnosis for various cutaneous manifestations. It is particularly crucial for emergency physicians caring for oncologic patients to be aware of cutaneous manifestations that herald severe and lifethreatening conditions to allow for quick recognition and initiation of the appropriate treatment. This chapter discusses the clinical presentation, pathophysiology, diagnosis, and management of these severe conditions organized by morphologic presentation (Table 34.1).

Maculopapular Eruptions

The differential diagnosis for maculopapular eruptions includes drug eruptions, viral exanthems, and, in the appropriate patient population, graft-versus-host disease.

Dermatology

Department of Dermatology, Weill Cornell Medical College, New York Presbyterian Hospital, New York, NY, USA e-mail: Joh9090@med.cornell.edu

Table 34.1 Cutaneous morphologies reviewed in this chapter

Maculopapular eruptions Localized erythema Generalized erythema (erythroderma) Vesicles and pustules Blistering diseases Purpuric (non-blanching) eruptions



Fig. 34.1 Maculopapular eruption: diffuse erythematous macules and papules in a patient with DRESS secondary to allopurinol

Drug Eruptions

A maculopapular (exanthematous, morbilliform) rash (MPR) is the most common form of adverse drug reaction in hospitalized patients, occurring in 57% of patients with a drug eruption [1]. The eruption consists of erythematous macules and papules scattered diffusely over the body (often sparing the face) that may coalesce.

DRESS (Drug Reaction with Eosinophilia and Systemic Symptoms)/DIHS (Drug-Induced Hypersensitivity Syndromes)

Clinical Manifestations

When a maculopapular drug rash presents with fever, lymphadenopathy, or facial edema, drug reaction with eosinophilia and systemic symptoms (DRESS)/drug-induced hypersensitivity syndrome (DIHS) should be considered, and a work-up for systemic involvement should be performed. The cutaneous eruption is typically a MPR and rarely presents with purpura, vesicles, or pustules (Fig. 34.1). The liver is the most common site of visceral involvement, but other systemic findings include interstitial nephritis, pneumonitis, myocarditis, arthritis, cytopenias, atypical lymphocytosis, thyroiditis, and cerebritis. The clinical manifestations typically begin 2–6 weeks after initial exposure to the causative medication necessitating a drug history of the past several months when DRESS is suspected.

Pathophysiology/Etiology

DRESS is a delayed type IV hypersensitivity reaction that may involve impaired pharmacokinetics leading to the accumulation of drug metabolites. Genetic predisposition given the association with specific HLA alleles, viral reactivation, and the release of cytokines, including interleukin 5, may play a role [2]. Common etiologies include the anticonvulsants, sulfonamides, allopurinol, dapsone, and antiretroviral medications. A few anticancer drugs have also been associated including chlorambucil and lenalidomide.

Diagnosis

Diagnosis is made based on clinical findings of a MPR plus evidence of internal organ involvement, most commonly eosinophilia and transaminitis. However, any organ system may be involved, and eosinophilia is not required for diagnosis (thus the term drug-induced hypersensitivity syndrome (DIHS) is sometimes used). Biopsy findings are not specific for DRESS and show overlap features with simple drug eruptions.

Treatment

Early discontinuation of the suspected medication is necessary. Systemic corticosteroids (typically prednisone 1-2 mg/ kg or equivalent) are the mainstay of treatment, and gradual taper with monitoring for flares is recommended to prevent relapse. The cutaneous and visceral manifestations may persist for weeks (rarely months), and patients should be monitored for late-onset thyroiditis and other autoimmune conditions.

Immune-Related Maculopapular Rash

Clinical Manifestations

Over 20% of patients treated with anti-cytotoxic T-lymphocyte-associated protein 4 (CTLA-4) or antiprogrammed death 1 (PD-1)/programmed death ligand 1 (PDL-1) and up to 72% with anti-CTLA-4/anti-PD-1 combination therapy develop immune-related cutaneous adverse events (ircAEs), most commonly pruritus and MPR [3, 4]. These ircAEs may be associated with prolonged progressionfree survival and overall survival [4]. Unlike MPR to traditional medications, those to immunotherapy may develop at any time during or months after completion of therapy, with a median onset of 62 days (range 1–1676 days) after initiation of therapy [4]. Cutaneous morphology of immunerelated MPR may be indistinguishable from that due to traditional medications or have overlap with lichenoid, psoriasiform, or eczematous rashes. Maculopapular rash severity is classified via the National Cancer Institute: Common Terminology Criteria for Adverse Events (CTCAE) with Grade 1, macules/papules covering <10% body surface area (BSA); G2, macules/papules covering 10–30% BSA or rash >30% BSA with or without mild symptoms; and G3, macules/papules covering >30% BSA with moderate or severe symptoms [5]. Immune-related MPR with or without systemic involvement have a longer median duration of 37.5 days (range 1–700 days) than simple drug eruptions to traditional agents [6].

Pathophysiology/Etiology

Immune checkpoint inhibitors block CTLA-4, PD-1, and PDL-1 generating durable antitumor responses in several cancers [3]. Side effects of these agents may be attributed to a persistently stimulated immune system and are thus termed "immune-related adverse events" (irAE) [3]. Elevations in eosinophil counts, IL-6, IL-10, and IgE have been associated with grade 3 ircAEs, suggesting a pathogenic or correlative role, as well as potential targets for supportive care intervention.

Diagnosis

Diagnosis is made based on clinical findings of a MPR and history of current or prior immune checkpoint inhibitor (ICI) therapy. Biopsy findings are not specific for immune-related maculopapular rash and show overlap features with traditional drug eruptions.

Treatment

Goal of treatment of immune-related MPR is to minimize systemic immunosuppressive treatments while maintaining patient on current dose of ICI. Management of G1 MPR consists of high-potency topical corticosteroid twice daily [4]. For G2 MPR, systemic corticosteroids (prednisone 0.5–1 mg/kg) may be added [5]. For G3 or intolerable G2 MPR, ICI should be interrupted until MPR severity decreases to G0-1; and dose modified as per protocol; oral corticosteroids (prednisone 0.5–1 mg/kg) or biologics (infliximab, tocilizumab) are administered [3]. Patients should be reassessed every 2 weeks until improvement in grade. If reaction worsens or does not improve, dose reduction or discontinuation per protocol may be necessary. For corticosteroid-resistant MPR, check serum levels of IL-6 and TNF- α to assess for eligibility for targeted therapy (e.g., tocilizumab, infliximab) [4].

Viral Exanthems

Viral exanthems often present with a maculopapular eruption that can mimic drug eruptions. They are commonly seen in the pediatric population and immunocompromised adults who are at higher risk of viral infection and reactivation.

Clinical Manifestations

Although viral exanthems typically present with a MPR, they may also be scarlatiniform (diffuse small papules giving a sandpaper quality to the skin), petechial (non-blanching), or urticarial. Other clinical findings include fever, myalgias, arthralgias, nausea, vomiting, diarrhea, and malaise. SARS-CoV-2 infection can present with a variety of skin manifestations including MPR, macular erythema, urticaria, vesicles, livedo, and purpuric eruptions which tend to appear concurrently or after other viral symptoms [7].

Pathophysiology/Etiology

Causes of viral exanthems include Epstein-Barr virus, cytomegalovirus, enterovirus, adenovirus, human herpesvirus types 6 and 7, parvovirus B19, measles (rubeola), rubella, and early human immunodeficiency virus, among others. It is not known how viruses cause skin rashes, though it is likely related to immune system activation.

Diagnosis

It is not always possible to determine the exact viral cause; however, serologies, viral polymerase chain reaction assays, heterophile antibodies, and the presence of atypical lymphocytosis may be helpful. Skin biopsy may be useful in differentiating drug eruptions from viral exanthems; however, findings can be nonspecific.

Treatment

Viral exanthems are often self-limited with resolution after 1–2 weeks. Treatment includes symptom management and supportive care.

Graft-Versus-Host Disease (GVHD)

Acute graft-versus-host disease (GVHD) commonly appears as a diffuse MPR that classically presents in the early posthematopoietic cell transplant period or after a reduction in immunosuppression (but can occur at any time after transplant). GVHD is most common after allogeneic hematopoietic stem cell transplant (alloHCT); however, it may be seen following autologous and solid organ transplant.

Clinical Manifestations

The skin is the most commonly affected organ in acute GVHD, presenting with a MPR. Individual lesions may be folliculocentric (centered upon a hair follicle) helping to distinguish from a drug or viral exanthem. A petechial component may be appreciated if the patient is thrombocytopenic. The liver and gastrointestinal systems are often also affected and, if severe, are associated with high morbidity and mortality. Chronic cutaneous GVHD has many clinical manifestations and can mimic lichen planus, lichen sclerosus, morphea,

and scleroderma. Engraftment syndrome presents around the time of neutrophil engraftment (ANC >500) with identical cutaneous manifestations as acute GVHD, as well as fever and evidence of capillary leak syndrome, and is more commonly seen after autologous transplant.

Pathophysiology/Etiology

In acute GVHD, alloHCT conditioning and damage to host tissues lead to the activation of host antigen-presenting cells. Donor T cells then proliferate in response to the activated antigen-presenting cells with subsequent activation of cytotoxic T lymphocytes and natural killer cells leading to tissue damage. Chronic GVHD is less well understood and is thought to involve donor T cells but also may involve B cells [8]. Risk factors associated with the development of GVHD include HLA incompatibility, unrelated donor, older age of recipient, peripheral blood stem cell source, and T-cell replete graft. In addition, the use of myeloablative-conditioning regimens is a specific risk factor for acute GVHD [9].

Diagnosis

The skin histology of acute GVHD shows varying degrees of keratinocyte necrosis, vacuolar changes at the dermalepidermal junction, and a lymphohistiocytic infiltrate in the upper dermis, but a definitive diagnosis requires clinical correlation. The histology of chronic GVHD typically reflects the clinical pattern of the skin manifestations. Several serum biomarkers are also under active investigation to aid in the diagnosis of GVHD.

Treatment

Topical steroids are used to treat limited cutaneous acute GVHD; however, extensive cutaneous disease and other organ involvements often require systemic corticosteroids. Alternative and adjunct treatments for acute and chronic GVHD include phototherapy, other immunosuppressives including tacrolimus, mycophenolate mofetil, Janus kinase (JAK) inhibitors, Bruton tyrosine kinase inhibitors, and extracorporeal photopheresis among others.

Localized Erythema

Cellulitis

Cellulitis is a superficial, diffuse inflammation of the cutaneous dermis and subcutaneous fat secondary to an infectious process. Underlying immunosuppression and disruption of the skin barrier contribute to the development of cellulitis. Table 34.2 Differential diagnosis for cellulitis in oncologic patients^a

Deep venous thrombosis
Thrombophlebitis
Lymphangitis
Venous stasis dermatitis
Allergic contact dermatitis
Lipodermatosclerosis
Erythema nodosum
Sweet syndrome
Radiation dermatitis and radiation recall
Deeper infection
Necrotizing fasciitis ^a
Osteomyelitis
Abscess
Pyomyositis

^aExpanded upon in text

Clinical Manifestations

Cellulitis presents with the acute onset of an erythematous, warm, tender plaque or plaques anywhere on the skin, usually over the lower extremities [10]. Cutaneous purpura may be present in the setting of thrombocytopenia or anticoagulation. Cutaneous edema can be severe leading to vesicle and blister formation. Cellulitis is almost always unilateral when located on an extremity. If findings are bilateral, an alternative diagnosis should be considered (Table 34.2). Patients are often afebrile, and increased white blood cell counts are seen in less than one-half of cases [10].

Pathophysiology

Cellulitis typically begins with organism entry through a disruption in the skin barrier especially in the setting of edema, trauma, ulceration, or a primary skin disorder, such as eczema or tinea pedis. The etiology is usually bacterial (most commonly *Streptococcus* genus followed by *Staphylococcus aureus*); however, cutaneous fungal infections including histoplasmosis and cryptococcosis can mimic bacterial cellulitis, especially in the immunocompromised host.

Diagnosis

Diagnosis is typically clinical. Skin biopsy is usually not helpful as histopathologic findings may be nonspecific, and tissue culture is positive for an organism in only about 20–30% of cases [10, 11]. However if cellulitis is worsening or not responding despite appropriate therapy, skin biopsy with tissue culture should be considered.

Treatment

Antibiotic therapy should be directed against streptococcal and staphylococcal organisms; however, broad-spectrum antibiotics including Gram-negative coverage are often warranted in the immunocompromised host. Adjunctive



Fig. 34.2 Necrotizing fasciitis: rapidly expanding erythema, purpura, and necrosis secondary to a polymicrobial deep soft tissue infection on the right leg

treatments including elevation and compression in the case of an involved extremity and treatment of concomitant skin conditions that disrupt the skin barrier should be initiated.

The differential diagnosis of cellulitis includes early necrotizing fasciitis, a rapidly progressive bacterial infection with necrosis of the deep subcutaneous tissue and fascia. Early infection presents with erythema and edema similar to cellulitis (often with pain out of proportion to exam); however, this is often quickly followed by the development of non-blanching purpura and hemorrhagic bulla that can progress to necrosis and gangrene within hours (Fig. 34.2). The most common etiology is group A *Streptococcus* though infections are often polymicrobial. Diagnosis is often made clinically, but imaging may demonstrate fascial thickening or air within the soft tissues. Treatment is emergent surgical debridement and broad-spectrum antimicrobials.

Toxic Erythema of Chemotherapy

Toxic erythema of chemotherapy (TEC) describes a spectrum of cutaneous eruptions that are thought to occur secondary to the cytotoxic effects of chemotherapy on the skin and sweat glands. Many terms have been used to describe these eruptions, including acral erythema, palmoplantar erythrodysesthesia, hand-foot syndrome, eccrine squamous syringometaplasia, Ara-C ears, and neutrophilic eccrine hidradenitis [12]. Skin findings in TEC include symmetric, erythematous, and purpuric (non-blanching) patches, which can be associated with erosions, bullae, and desquamation (Fig. 34.3) frequently noted in intertriginous areas, genital, overlying joints, and acrally. Patients often describe burning or pruritus.

TEC has been attributed to cytarabine, doxorubicin, 5-fluorouracil, capecitabine, methotrexate, bleomycin, carboplatin, cisplatin, etoposide, gemcitabine, receptor tyrosine



Fig. 34.3 Toxic erythema of chemotherapy: erythema and blisters secondary to sorafenib

kinase inhibitors, cyclophosphamide, and melphalan, among others [12]. Diagnosis is often made clinically. TEC is typically self-limited, and treatment is largely supportive and includes topical steroids, analgesics, and emollients. Rarely systemic steroids are needed in severe cases. Prevention includes dose reduction and lengthening dose intervals.

Leukemia Cutis

Leukemia cutis is a skin eruption that results from cutaneous infiltration of malignant cells in the setting of leukemia, most commonly acute myeloid leukemia. Leukemia cutis classically presents with single or multiple raised red-purple (plum-colored) papules or nodules that can arise in any location [13]. Leukemia cutis may be the initial presenting manifestation of leukemia [14]. Skin biopsy should be performed to confirm the diagnosis and rule out clinical mimics including cutaneous infection and Sweet syndrome (acute febrile neutrophilic dermatosis). Management is aimed at treatment of the underlying leukemia. The development of leukemia cutis typically portends a poorer prognosis, with the exception of congenital leukemia with leukemia cutis [14].

Angioedema

Angioedema is the swelling of the deep dermis, subcutaneous, and submucosal tissues, most commonly on the eyelids, lips, and genitalia. It can involve the tongue and pharynx and, in such cases, become life-threatening.

Clinical Manifestations

The affected area is edematous and tender with the surface appearing normal or slightly pink. Angioedema can occur alone or simultaneously with hives. Patients with angioedema or hives must be monitored for signs and symptoms of anaphylaxis with evidence of respiratory compromise, hypotension, and shock.

Pathophysiology/Etiology

The etiology of angioedema is determined by the clinical manifestations and is critical for appropriate treatment. Angioedema that occurs with hives is due to an immediate type I hypersensitivity reaction mediated by IgE and mast cell degranulation. It can be triggered by infection, foods, and drugs or may be idiopathic. Angioedema that occurs without hives may be related to an inherited mutation or acquired in the presence of an underlying malignancy or autoimmune disorder. This pathway is driven by deficient or dysfunctional C1q esterase inhibitor, a critical enzyme in the complement and fibrinolytic cascade, leading to increased levels of bradykinin with resultant vasodilation and edema.

Diagnosis

Diagnosis of angioedema is often clinical; however, in the case of hereditary or acquired angioedema, low complement 4 (C4) level is seen both during and between attacks. C1q level is also low in acquired angioedema, helping to distinguish it from hereditary cases.

Treatment

Angioedema due to a type I hypersensitivity reaction may be treated with aggressive antihistamines and discontinuation of the offending agent if known. If there is concern for anaphylaxis, intramuscular or intravenous epinephrine should be used promptly. Recurring cases may require leukotriene receptor antagonists, immunosuppressants, and biologics [15].

Treatment for acute attacks in hereditary or acquired angioedema includes fresh frozen plasma, C1 inhibitor concentrate, kallikrein inhibitors, and bradykinin receptor antagonists [16].

Erythroderma

Erythroderma describes diffuse erythema of most if not the entire cutaneous surface often with exfoliative shedding of the skin. The differential diagnosis includes drug-induced, primary skin disorders, infectious etiologies, and paraneoplastic causes, among others (Table 34.3).

Table 34.3 Differential diagnosis for erythroderma

Drug-induced
Primary skin disorder
Psoriasis
Atopic dermatitis
Allergic contact dermatitis
Chronic actinic dermatitis
Pemphigus foliaceous
Infection
Toxic shock syndromes ^a
Generalized dermatophytosis
"Norwegian" crusted scabies
Viral exanthem
Neoplastic
Mycosis fungoides/Sézary syndrome ^a
Paraneoplastic phenomenon
Graft-versus-host disease
Others (rare): nutritional deficiencies, Kawasaki disease, cutaneous
mastocytosis
^a Expanded upon in text

Toxic Shock Syndrome

Toxic shock syndrome (TSS) is an acute, life-threatening infection due to a toxin-producing strand of *Staphylococcus* or *Streptococcus*. Patients with underlying chronic medical conditions including cancer are at higher risk for TSS.

Clinical Manifestations

In both staphylococcal and streptococcal TSS, patients become acutely ill with high fever, hypotension, and evidence of multiorgan system involvement. Influenza-like symptoms including chills, myalgias, headache, vomiting, and diarrhea are common. Cutaneous findings are relatively nonspecific including subtle, diffuse, blanchable erythema as well as edema and erythema of the palms and soles followed by prominent desquamation within 1–2 weeks. Mucosal findings include conjunctival erythema, anogenital erythema, and a strawberry tongue.

Pathophysiology/Etiology

In TSS, specific strands of *Staphylococcus* and *Streptococcus* produce toxins that act as superantigens leading to a wide-spread T-cell activation and cytokine release that cause fever, capillary leakage, and hypotension. Staphylococcal TSS is most often associated with focal infections including surgical wound infections, burns, osteomyelitis, sinusitis, septic arthritis, and tampon use in menstruating women [17]. Streptococcal TSS is more often seen in connection with bacteremia, cellulitis, or necrotizing fasciitis [18].

Diagnosis

Both staphylococcal and streptococcal TSS have specific diagnostic criteria required for diagnosis (Tables 34.4 and 34.5).

Fever
Hypotension
Diffuse erythroderma
Desquamation (1-2 weeks after illness onset)
Three or more of the following:
Renal dysfunction
Gastrointestinal: vomiting or diarrhea
Hepatic dysfunction
Hematologic dysfunction
Severe myalgias or elevated creatinine phosphokinase
Altered mental status
Mucous membranes erythema
Negative results of following tests, if obtained:
Blood, throat, or cerebrospinal fluid cultures for another pathogen

Serologic tests for Rocky Mountain spotted fever, leptospirosis, measles

Data from the Centers for Disease Control. Wharton et al. [60] and Centers for Disease Control and Prevention [61]

 Table 34.5
 Diagnostic
 criteria
 for
 streptococcal
 toxic
 shock

 syndrome

 <t

Isolation of group A Streptococcus
Hypotension
Two or more of the following:
Pulmonary dysfunction
Liver dysfunction
Renal dysfunction
Coagulopathy
Erythroderma of the skin
Soft tissue necrosis (e.g., necrotizing fasciitis)

Data from: The Working Group on Severe Streptococcal Infections. [62]

Positive blood cultures are more frequently seen in streptococcal TSS than in staphylococcal TSS.

Treatment

Treatment includes rapid introduction of intravenous antibiotics against *Staphylococcus* and *Streptococcus*. Clindamycin has direct antitoxin properties and has been shown to improve patient outcomes in TSS [19]. The source of infection should be investigated and may be occult. Intravenous immunoglobulin (IVIG) and corticosteroids may be beneficial in severe and refractory cases [20, 21].

Mycosis Fungoides and Sézary Syndrome

Mycosis fungoides (MF) and Sézary syndrome (SS) are types of cutaneous T-cell lymphoma (CTCL), and both can present with erythroderma. MF is a T-cell lymphoma with



Fig. 34.4 Erythroderma: diffuse erythematous scaly plaques with small areas of sparing secondary to extensive mycosis fungoides

initial presentation in the skin but with potential involvement of the lymph nodes, blood, and internal organs. SS is a distinctive type of CTCL with leukemic involvement of malignant T cells.

Clinical

MF typically presents with patches and plaques and may progress to tumors or erythroderma (Fig. 34.4). SS generally presents with erythroderma often in the setting of lymphadenopathy and indurated facial features from the infiltration of malignant cells [22].

Etiology

The etiology of cutaneous T-cell lymphoma is not entirely known. Hypotheses include chronic stimulation of circulating skin-homing lymphocyte subsets that cause transformation to a monoclonal population.

Diagnosis

Atypical lymphocytes within in the epidermis on skin biopsy are diagnostic of MF, though there may be overlap features with other chronic dermatoses. Diagnosis of SS requires the involvement of a clonal neoplastic T-cell population in the skin, lymph nodes, and blood.

Treatment

Patch stage MF may be treated with skin-targeted remedies such as topical steroids, topical nitrogen mustard, topical bexarotene, and phototherapy. Advanced MF presenting with erythroderma or systemic involvement and SS requires systemic therapies such as extracorporeal photopheresis, systemic retinoids, interferons, monoclonal antibodies, and chemotherapy regimens. Table 34.6 Differential diagnosis for common causes of vesicular/ pustular eruptions in oncologic patients

Drug
Acute generalized exanthematous pustulosis (AGEP)
Acneiform eruption secondary to epidermal growth factor receptor (EGFR) inhibitors
Steroid-induced acneiform eruption
Drug-induced autoimmune blistering disease (especially
secondary to immunotherapy)
Inflammatory
Pustular psoriasis
Miliaria
Allergic/irritant contact dermatitis
Neutrophilic dermatoses
Infectious
Viral
Herpes simplex virus ^a
Varicella-zoster virus ^a
Coxsackie virus (hand, foot, and mouth disease)
SARS-CoV-2 (rare)
Bacterial
Bacterial folliculitis
Ecthyma ^a
Others: rickettsialpox, nocardiosis, listeriosis
Fungal
Disseminated candidiasis ^a
Disseminated opportunistic fungal infection
Atypical mycobacteria
^a Expanded upon in text

xpanded upon in text

Vesicles and Pustules

The differential diagnosis for vesicular and/or pustular eruptions in oncologic patients is broad and includes infectious causes, inflammatory disorders, and drug reactions (Table 34.6).

Drug-Induced Autoimmune Blistering Disease

Immune-Related Bullous Pemphigoid-Like Eruption

Clinical Manifestations

Immune-related bullous pemphigoid-like (irBP) eruption due to ICI occurs in 1-5% of patients treated with anti-PD1 or anti-CTLA-4 and may develop during or months after completion of ICI with a median 190-298.5 days (range 31-2380 days) after initiation of therapy [3, 4, 23]. Development of irBP is associated with improved tumor response after PD-1 therapy [24]. Cutaneous morphology of irBP may be indistinguishable from idiopathic BP. Bullous dermatitis severity is classified via CTCAE with G1, asymptomatic, blisters covering <10% BSA; G2, blisters covering 10-30% BSA, painful blisters, limiting instrumental ADL; G3, blisters covering >30% BSA, limiting self-care ADL; and G4, blisters covering >30% BSA, associated with fluid or electrolyte abnormalities, ICU care, or burn unit indicated [5]. Median duration of irBP is 97.5 days (range 13-390 days) [4].

Pathophysiology/Etiology

The exact pathomechanism of irBP is not yet established. In idiopathic (non-drug-related) bullous pemphigoid, BP180, a transmembrane glycoprotein in the skin, serves as an autoantigen targeted by antibodies leading to dermal-epidermal separation and blister formation. BP180 is expressed in melanoma and non-small cell lung cancer, possibly serving as the target of antitumor antibodies that may concurrently bind to cutaneous BP180 [23].

Diagnosis

Diagnosis is made based on clinical findings of a tense bullae or urticarial plaques and history of current or prior immune checkpoint inhibitor (ICI) therapy. Differential diagnosis for irBP includes linear IgA bullous dermatosis (LABD), bullous lichen planus, lichen planus pemphigoides, bullous erythema multiforme, and epidermolysis bullosa acquisita. Serologic testing by ELISA for circulating autoantibodies against BP180 and BP230 may be used to confirm diagnosis, correlate with disease severity, or monitor response to treatment. Histology will reveal a subepidermal bullous dermatitis with eosinophils, and direct immunofluorescence (DIF) shows linear deposition of IgG and C3 at the basement membrane zone of the dermoepidermal junction [3]. Histologic and direct immunofluorescence findings will not differentiate between irBP and idiopathic BP, but will distinguish from the aforementioned bullous conditions which may also be immunotherapy associated.

Treatment

Goal of treatment of immune-related BP is to minimize systemic immunosuppressive treatments while maintaining the patient on the current dose of ICI. Management of G1 irBP consists of high-potency topical corticosteroid twice daily [2]. For G2 irBP, hold ICI until severity decreases to grade 0-1, and add systemic corticosteroids (prednisone 0.5-1 mg/ kg) [5]. For G \geq 3 or intolerable G2 irBP, oral corticosteroids (prednisone 0.5-1 mg/kg) and rituximab (+/- IVIG [4]) are added [3]. Patients should be reassessed every 2 weeks until improvement in grade. If reaction worsens or does not improve, dose reduction or discontinuation per protocol may be necessary.

Viral Infections

Varicella-Zoster Virus

Varicella-zoster virus (VZV) from the *Herpesviridae* family is the cause of varicella (chickenpox) and zoster (shingles). Herpes zoster is a common infection seen in the oncologic patient.

Clinical Presentation

Varicella begins with mild fever, malaise, and myalgias followed by an eruption of 2–4 mm clear vesicles with a red rim. Over the course of several days, the vesicles become pustular and often form a prominent central hemorrhagic crust. Varicella in adults and immunocompromised patients may be associated with higher risk of morbidity and mortality, often with more extensive crusting and risk of internal organ involvement.

Herpes zoster is the reactivation of latent VZV. Zoster initially presents with a prodrome of burning, tingling, and pruritus followed by the development of grouped vesicles on an erythematous base in a dermatomal distribution. When certain dermatomes are affected, extracutaneous complications can occur including ocular complications, facial paralysis, loss of taste, deafness, and temporary inability to urinate or defecate. In the immunocompromised, pain and post-herpetic neuralgia may be more severe [25].

Disseminated zoster is defined as more than 20 vesicles outside the primary or contiguous dermatome (Fig. 34.5). Visceral involvement including pulmonary, hepatic, and central nervous system can occur in approximately 10% of immunocompromised patients [26].

Pathophysiology/Etiology

Varicella is transmitted through airborne droplets or direct contact with vesicular fluid. After varicella infection, the virus travels to the dorsal root ganglion where it remains latent until reactivation. It is the reactivation of latent VZV that causes zoster (shingles), which may occur spontaneously or in the setting of stress, fever, local trauma, immunosuppression, or radiation.

Diagnosis

The diagnosis can be made clinically based on typical lesional morphology. Polymerase chain reaction (PCR) from lesional fluid has become the preferred method for confirmatory testing based on higher sensitivity and rapid results (often within 24 h). If not available, a Tzanck smear or direct fluorescence antibody (DFA) assay or viral culture can be used [27].



Fig. 34.5 Disseminated zoster: scattered vesicles on an erythematous base diffusely on the back with grouped vesicles in a dermatomal distribution on right mid-back

Treatment

Antivirals can be used for varicella and zoster in immunocompetent patients to decrease the duration and severity. Intravenous acyclovir is recommended for varicella in immunocompromised individuals and disseminated zoster. Postexposure prophylaxis with varicella-zoster immunoglobulin is recommended for immunocompromised individuals and nonimmune pregnant women.

Eczema Herpeticum (Kaposi's Varicelliform Eruption)

Eczema herpeticum is the cutaneous dissemination of herpes simplex virus (HSV) in areas of a preexisting dermatitis, such as atopic dermatitis (eczema), mycosis fungoides, or other skin conditions with impaired skin barrier.

Clinical Presentation

Eczema herpeticum appears clinically as discrete 2–3 mm punched-out erosions and circular hemorrhagic crusts con-



Fig. 34.6 Eczema herpeticum: diffuse 2–3 mm punched-out erosions within an area of eczema on the foot

centrated in areas of dermatitis (Fig. 34.6). Occasionally intact grouped vesicles or vesiculopustules may be seen, and the lesions may be superinfected with bacteria. Patients may have associated fever, malaise, and lymphadenopathy.

Pathophysiology/Etiology

Transmission of HSV1 is typically through direct contact with contaminated saliva, while HSV2 is transmitted through sexual contact. The virus then replicates at the site of infection and travels to the dorsal root ganglia, where it establishes latency until reactivation. Upon reactivation, the virus is able to spread via impaired skin leading to widespread involvement.

Diagnosis

As above, the preferred diagnostic technique is lesional PCR, though viral culture, DFA, and Tzanck smears may also be used.

Treatment

Treatment includes the use of antiviral therapy for 10–14 days, especially if immunocompromised, until all



Fig. 34.7 Ecthyma: large pustules with central necrosis and crusting secondary to *S. aureus* bacteremia

lesions are crusted over. Severe cases may require hospitalization with empiric intravenous acyclovir, while diagnostic studies are pending. Clinicians should also have a high suspicion for bacterial superinfection (especially with staphylococcal species) and low threshold to start concurrent antibiotics.

Bacterial Infections

Ecthyma

Ecthyma is an ulcerative bacterial skin infection that can be localized or widespread with systemic manifestations. Ecthyma gangrenosum is a specific term for ecthyma skin lesions secondary to bacteremia with *Pseudomonas aeruginosa*.

Clinical Presentation

Ecthyma initially begins as single or multiple vesiculopustules that enlarge over several days. Lesions then ulcerate and develop central necrotic adherent crusts (Fig. 34.7). When multiple lesions are present, patients may have fever, chills, malaise, and, sometimes, hypotension and shock. Bacteremia should be strongly suspected.

Pathophysiology/Etiology

Ecthyma is secondary to a localized skin infection or secondary to bacteremia with cutaneous seeding. Ecthyma may be due to Gram-positive organisms including *Streptococcus* and *Staphylococcus* species and Gram-negative organisms such as *Pseudomonas*.

Diagnosis

The diagnosis can be made based on clinical appearance and Gram stain and culture of the purulent base. Skin biopsy and tissue culture can confirm the diagnosis and organism. Blood cultures should be ordered especially when multiple lesions are present.

Treatment

Treatment includes systemic antibiotics and wound care with soaking and gentle debridement of adherent crusts and topical antibiotic ointment. Hospitalization and intravenous antibiotics are indicated when multiple lesions are present or in immunocompromised patients.

Fungal Infections

Disseminated Candidiasis

Disseminated candidiasis is an infection most often seen in the immunocompromised host and can affect any organ system including the skin.

Clinical Manifestations

Cutaneous manifestations of disseminated candidiasis include pustules and scattered erythematous macules, papules, or nodules, often with a pale center. Other presentations include hemorrhagic bulla and purpura, especially in the setting of thrombocytopenia. Individuals are typically illappearing with tachycardia, hypotension, and fever.

Pathophysiology/Etiology

Candidal sepsis commonly occurs from *Candida* that has colonized the gastrointestinal tract or skin. *Candida albicans* is a common etiology of disseminated candidiasis, but *C. glabrata* and *C. tropicalis* can also be seen. *C. tropicalis* is more likely to produce cutaneous lesions and is common in patients with leukemia [28].

Diagnosis

Diagnosis can be established through skin biopsy and tissue culture or potassium hydroxide (KOH) preparation of purulent material. Budding yeast and pseudohyphae in the dermis are seen on skin biopsy. Evaluation also includes blood cultures, which may be negative, and evaluation for other organ involvement.

Treatment

In non-neutropenic patients, disseminated candidiasis can be treated with fluconazole. In neutropenic patients, amphotericin B, caspofungin, or voriconazole should be used. Any foci of infection such as lines and catheters should be removed.

Blistering Diseases

Stevens-Johnson Syndrome and Toxic Epidermal Necrolysis

Stevens-Johnson syndrome (SJS) and toxic epidermal necrolysis (TEN) are severe mucocutaneous reactions that are typically triggered by medications. SJS/TEN may be more frequent in the oncologic population with one study showing a high incidence of TEN (6%) in bone marrow transplant patients, though cases of TEN-like acute graft-versus-host disease (GVHD) were included [29].

Clinical Manifestations

SJS/TEN typically presents 1–3 weeks after initiation of the offending medication with blisters centered on non-blanching macules or atypical target lesions characterized by two distinct zones: a dark red center surrounded by a pale red outer ring (Fig. 34.8). The majority of immune checkpoint inhibitor (ICI)-associated SJS/TEN-like reactions begin after the first or second infusion of ICI; however, it may occur even 140 days after initiation of therapy or after discontinuation [30, 31]. Lesions tend to start proximally on the trunk and face and then spread distally. Individual lesions can rapidly coalesce followed by cutaneous necrosis and epidermal sloughing that can become widespread. ICI-associated SJS/TEN-like reactions may be preceded by days to months of slowly evolving maculopapular, psoriasiform, or lichenoid



Fig. 34.8 SJS: coalescing blisters centered on non-blanching macules on the chest



Fig. 34.9 (a) Generalized dusky red maculopapular eruption with scattered areas of erosions and hemorrhagic crusting, (b) cutaneous and mucosal lips with hemorrhagic mucositis, (c) confluent oral erosions with white fibrinous pseudomembranes

rashes, before developing denudation and mucosal involvement.

SJS/TEN exist on a spectrum defined by percentage body surface area (BSA) of epidermal detachment: SJS defined as <10% BSA, SJS/TEN as 10–30%, and TEN as >30%. ICIassociated SJS/TEN-like reactions severity is additionally graded via CTCAE starting with G3, skin sloughing covering <10% BSA with associated signs (e.g., erythema, purpura, epidermal detachment, and mucous membrane detachment); G4, skin sloughing covering 10–30% BSA with associated signs (e.g., erythema, purpura, epidermal detachment, and mucous membrane detachment); and G5, death [5] (Fig. 34.9).

Erythema multiforme (EM) was previously considered by many to exist on the spectrum of SJS and TEN but has more

recently been classified as a separate entity with distinctive skin findings and etiology and a good prognosis [32].

Typically multiple mucosal sites (oral, ocular, and/or anogenital) are involved with erosions, ulcerations, and hemorrhagic crusting [33] (Fig. 34.10). In severe cases, gastrointestinal (GI) and pulmonary involvement occur. Patients are often febrile and can display signs of shock. Associated organ system involvement is frequently reported including hepatitis, acute renal failure, myocarditis, and bone marrow suppression. One study found the mean adjusted mortality rates between 2009 and 2012 in the United States were 4.8% for SJS, 19.4% for SJS/TEN, and 14.8% for TEN [34].

A validated scoring system has been developed to assess the severity of illness and predict mortality in SJS/TEN



Fig. 34.10 Mucosal findings in SJS: erosions and hemorrhagic crusting

 Table 34.7
 SCORTEN severity of illness score in Stevens-Johnson syndrome and toxic epidermal necrolysis (SJS/TEN)

SCORTEN Score of toxic epidermal necrosis

(Table 34.7). However, with advances in wound care and treatment, the SCORe of Toxic Epidermal Necrosis (SCORTEN) may over-estimate mortality [35].

Pathophysiology

The pathophysiology of SJS/TEN is likely multifactorial involving cytotoxic T cells and soluble mediators including perforin, granzyme B, granulysin, and Fas-Fas ligand (FasL).

 Table 34.8
 Differential diagnosis for Stevens-johnson syndrome and toxic epidermal necrolysis (SJS/TEN)

Erythema multiforme^a Staphylococcal scalded skin syndrome Pemphigus vulgaris/paraneoplastic pemphigus^a TEN-like acute graft-versus-host disease^a Linear IgA bullous dermatosis Generalized bullous fixed drug eruption ^aExpanded upon in text

Certain populations have specific human leukocyte antigen (HLA) types that are associated with a higher risk of development of SJS/TEN indicating a genetic component. Hundreds of medications have been associated with SJS/ TEN, but the most common include allopurinol, aromatic anticonvulsants and lamotrigine, sulfa antibiotics, nevirapine, and nonsteroidal anti-inflammatory drugs (NSAIDs). In addition, many anticancer drugs have been associated with SJS/TEN including lenalidomide, thalidomide, docetaxel, and imatinib [36] and with SJS/TEN-like reactions including anti-CTLA-4, anti-PD-1, and anti-PDL-1 inhibitors [30]. The mechanism of SJS/TEN-like reactions is not yet established. Blockade of homeostatic PD-1/PD-L1 signaling, which maintains peripheral tolerance in the skin, may allow autoreactive CD8+ T cells targeting keratinocytes displaying self-antigens to become activated and proliferate leading to development of SJS/TEN-like reactions [37].

Diagnosis

Diagnosis can often be made clinically based on acute onset of blisters, targetoid (target-like) lesions, and epidermal detachment. Skin biopsy confirms the diagnosis and shows full-thickness epidermal necrosis. Where available, frozen sectioning of skin biopsy samples allows rapid diagnostic confirmation within hours. Direct immunofluorescence should also be done to rule out other etiologies of blistering disease, which is especially important for patients with history of ICI therapy given their predisposition for immunobullous disease (Table 34.8). SJS/TEN and ICI-associated SJS/TEN-like eruptions have overlapping histologic features.

Diagnosis may be more complicated in the setting of oncologic patients especially those who have undergone stem cell transplantation as severe acute GVHD can mimic SJS/TEN. Both diseases are at least partially mediated by cytotoxic T cells that target keratinocytes. Skin biopsies may be indistinguishable though the presence of eosinophils may be suggestive of GVHD.

Treatment

The most important interventions that have consistently been shown to improve survival in SJS/TEN is prompt discontinuation of the offending medication and transfer to a burn unit, intensive care unit, or specialized care center. Supportive care consists of fluid and electrolyte balance, wound care, and monitoring for/early treatment of infection. An ophthalmology consult should be called whenever SJS/ TEN is suspected. A urology and/or gynecology consult should also be considered. The use of prophylactic antibiotics is not recommended as no survival advantage has been shown [38].

A recent shift toward the use of cyclosporine and the tumor necrosis factor inhibitor etanercept has changed the landscape for treatment in patients with SJS/TEN. A 2018 meta-analysis of over 250 patients with SJS/TEN found that treatment with cyclosporine was associated with an approximately 70% decrease in mortality [39]. Typical recommended dose is 3 to 5 mg/kg/d divided bid for 7-10 days [39]. ICI-associated SJS/TEN-like eruptions may require longer therapeutic courses due to persistent immune activation and longer half-life of ICI when compared to traditional drug culprits. A 2018 randomized trial of 91 patients comparing etanercept versus systemic steroids showed improved clinical outcomes including decreased SCORTEN predicted mortality, decreased side effects, and more rapid reepithelialization in the etanercept group [40]. While these results are encouraging, randomized, prospective trials are needed.

The use of systemic steroids in SJS/TEN (including recommended route of administration, dosage, and length of administration) remains controversial [35, 41-45]. While a recent 2017 meta-analysis [41] suggested decreased mortality rate, others have shown either no benefit or increased infection risk and length of hospital stay [35]. Some studies suggest a benefit especially if used earlier in the disease course, and at high dosages however, benefits must be weighed against increased infection risk and potential for delayed re-epithelialization [42, 43]. Intravenous immunoglobulin use in SJS/TEN is similarly controversial with some studies suggestive of benefit at higher dosages (over 2 gm/ kg) [44], while a meta-analysis from the EuroSCAR group did not show survival benefit [45]. Despite this lack of clear supportive data, IVIG is still widely used for SJS/TEN in the United States [35]. Another important treatment consideration is the use of steroids and IVIG concurrently which also warrants additional study [35].

Paraneoplastic Pemphigus

Paraneoplastic pemphigus (PNP) is an autoimmune mucocutaneous blistering disease associated with an underlying neoplasm. PNP has been described in association with lymphoproliferative neoplasms including chronic lymphocytic leukemia, non-Hodgkin's lymphoma, and Castleman's disease, though solid organ tumors including thymomas have also been reported [46].

Clinical Manifestations

PNP is characterized by painful mucosal ulcerations and a polymorphous skin eruption with an associated neoplasm. Mucosal involvement typically affects the oral mucosa, especially the lips; however, the conjunctiva, anogenital region, nasopharynx, and esophagus may also be involved. Skin manifestations typically appear later and are varied including nonspecific erythematous papules, target-like lesions, and blisters. Internal organ involvement has also been reported including pulmonary (classically bronchiolitis obliterans), thyroid, renal, and gastrointestinal tract.

Pathophysiology

Autoantibodies against plakins (periplakin, envoplakin) are diagnostic of PNP. Plakins are important proteins found in hemidesmosomes and desmosomes, which serve as vital structures in keratinocyte adhesion to other keratinocytes and to the underlying basement membrane. In some cases, these autoantibodies have been shown to be produced directly by the associated neoplasm [47]. Additional autoantibodies involved in PNP are desmoplakins 1 and 2 and plectin.

Diagnosis

Currently no single established set of diagnostic criteria exists for PNP. Most proposed definitions include mucosal involvement, detection of autoantibodies via direct and/or indirect immunofluorescence antibody testing against envoplakin and/or periplakin, and the presence of an underlying neoplasm.

Treatment

PNP often improves after the removal or treatment of the underlying neoplasm. Rituximab is recommended as firstline treatment for PNP [48]. Other treatments including prednisone, cyclosporine, cyclophosphamide, and IVIG have also shown efficacy [46, 49].

Purpuric (Non-blanching) Eruptions

Purpuric eruptions describe non-blanching skin lesions secondary to hemorrhage into the skin. Purpuric lesions may be flat (macular purpura), small and raised (palpable purpura), or larger and netlike (retiform purpura). It is important to recognize the features of macular purpura, palpable purpura, and retiform purpura as the differential diagnosis varies based on these morphologic differences.

34 Dermatology

Table 34.9 Differential diagnosis for macular purpura

Platelet-related
Low platelets
Immune thrombocytopenic purpura (ITP)
Thrombotic thrombocytopenic purpura/hemolytic uremic syndrome (TTP/HUS)
Disseminated intravascular coagulation (DIC)
Bone marrow failure
Drug-induced thrombocytopenia
Cirrhosis
Abnormal platelets
Congenital or hereditary platelet dysfunction
Nonsteroidal anti-inflammatory drugs
Renal disease
Thrombocytosis
Non-platelet-related
Trauma/Valsalva
Infections (Rocky Mountain spotted fever, parvovirus B19, disseminated strongyloidiasis)
Capillary fragility (actinic damage, Ehlers-Danlos syndrome)
Anticoagulant medications
Vitamin K deficiency
Vitamin C deficiency

 Table 34.10
 Differential diagnosis for palpable purpura (cutaneous small vessel vasculitis)

Inflammatory
Connective tissue disease-associated vasculitis
Mixed type II and III cryoglobulinemia
Henoch-Schonlein purpura (IgA vasculitis)
ANCA+ vasculitis
Infections (most commonly Streptococcus, HIV, hepatitis,
tuberculosis)
Medications
Neoplastic (leukemic vasculitis, paraneoplastic phenomenon)
Idiopathic

Macular Purpura

Macular purpura describes flat areas of purpura of varying sizes. Lesions may be small (petechiae) or larger (ecchymoses). Macular purpura typically indicates hemorrhage into the skin secondary to low or dysfunctional platelets or vessel wall fragility in the absence of inflammation (Table 34.9).

Palpable Purpura

Palpable purpura describes small, raised, non-blanching lesions most commonly found on the lower extremities. Palpable purpura is the classic skin manifestation for cutaneous small vessel vasculitis (Table 34.10).



Fig. 34.11 Retiform purpura: netlike pattern of cutaneous purpura with central necrosis on the abdomen

Retiform Purpura

Retiform purpura describes cutaneous lesions that have a netlike or stellate (starlike) pattern of purpura often with central necrosis or ulceration, reflecting damage to larger vessels with resultant cutaneous ischemia and hemorrhage (Fig. 34.11). Damage to the vessel may occur through either infiltration of the vessel wall or occlusion of the vessel lumen (Table 34.11).

Acute Meningococcemia

Meningococcemia, a bloodstream infection with *Neisseria meningitidis*, is a rapidly progressive disease with a fatality rate of 7–11% [50]. Though its incidence is decreasing with increased vaccination in well-developed countries, its severity makes rapid recognition of paramount importance, especially in the immunocompromised host.

Clinical Manifestations

Acute infection results in a constellation of symptoms including high fever, myalgias, neck pain or stiffness, and headache. Skin findings typically manifest with a diffuse petechial rash that rapidly progresses to retiform purpura with a central "gun-metal gray" color and necrosis. Hypotension and shock may develop, and patients should be monitored for disseminated intravascular coagulation.

Pathophysiology/Etiology

Acute meningococcemia is caused by transmission via droplet of *Neisseria meningitidis*, a Gram-negative coccus. Disease typically develops 2 weeks after colonization of the pharyngeal mucosa. Cutaneous lesions of retiform purpura

Table 34.11 Differential diagnosis for retiform purpura

Vessel wall infiltration
Vasculitis
Septic vasculitis (bacterial, angioinvasive fungal)
Mixed type II and III cryoglobulinemia
Connective tissue disease-associated vasculitis
ANCA+ vasculitis
Leukemic vasculitis
Polyarteritis nodosa
Drug-induced vasculitis
Deposition (calciphylaxis, oxalosis)
Vessel lumen occlusion
Thrombotic
Abnormal coagulation
Hypercoagulable state (acquired or hereditary)
Warfarin-induced skin necrosis ^a
Disseminated intravascular coagulation/purpura fulminans
COVID-19-associated coagulopathy
Platelet plugging
Heparin-induced thrombocytopenia (HIT) ^a
Thrombotic thrombocytopenic purpura/hemolytic uremic
syndrome (TTP/HUS)
Paroxysmal nocturnal hemoglobinuria
Myeloproliferative disorders (essential thrombocytosis)
RBC occlusion (sickle cell, hereditary spherocytosis)
WBC occlusion (intravascular lymphoma)
Cold-related (type I cryoglobulinemia, cryofibrinogenemia)
Embolic: (septic, cholesterol, cardiac, air, fat emboli)

^aExpanded upon in text

result from bacterial proliferation within the blood vessels creating vascular occlusion.

Diagnosis

Diagnosis of acute meningococcemia is clinical and should be suspected in any febrile patient with retiform purpura especially in the setting of headache and neck pain or stiffness. Treatment should be initiated prior to results of diagnostic testing to prevent rapid spread of infection. Cerebrospinal fluid (CSF) culture is superior to blood culture and positive in 90% of patients. PCR-based techniques on CSF approach a sensitivity of 100% [50]. Skin biopsy with tissue culture can be helpful to confirm the diagnosis.

Treatment

Neisseria meningitidis is highly sensitive to high-dose intravenous penicillin G, ceftriaxone, and cefotaxime, and treatment should be initiated within 1 h of presentation. Rifampin should be given to close contacts, and the vaccine is available for high-risk groups [51]. Adult oncologic patients are not routinely vaccinated unless a risk factor, such as previous splenectomy, is present.

Opportunistic Fungal Infections

Infections from the opportunistic dermatomycoses including aspergillosis, fusariosis, cryptococcosis, and zygomycosis

can range from localized cutaneous infection to disseminated infection with multiorgan system involvement. Leukemia and lymphoma patients are most at risk for deep fungal infections when neutropenic.

Clinical Manifestations

Disseminated fungal infections can present with a wide range of cutaneous lesions including retiform purpura, papulonecrotic lesions, pustules, and hemorrhagic bulla [52]. *Cryptococcus* may present with umbilicated papules that resemble molluscum contagiosum. Patients are typically febrile and appear acutely ill.

Pathophysiology/Etiology

Systemic mycoses may begin with primary skin infection or spread to the skin from a distant site of infection. Disruption of skin barrier and mucosa from burns, trauma, and indwelling catheters, along with impaired immunity, are predisposing factors.

Diagnosis

Diagnosis can be established with skin biopsy, tissue culture, and microscopic examination of lesional fluid. Serum assays for components of the fungal cell wall, including 1,3-beta-Dglucan and galactomannan, can be used to screen for invasive fungal infection. Cryptococcal antigen is measurable in serum and correlates with fungal burden. Work-up should include chest X-ray and sputum culture to evaluate for pulmonary involvement.

Treatment

Expeditious treatment of deep fungal infection is key, and delay in treatment by as little as 2 h has shown to correlate with increasing mortality [53]. For many of the disseminated opportunistic mycoses, treatment is with intravenous amphotericin B. Voriconazole is first-line treatment for invasive aspergillosis. Prognosis is poor in disseminated disease but can be improved if neutropenia improves. Prophylactic treatment (agent of choice depends on the underlying malignancy, the length of expected neutropenia, etc.) may help reduce the risk of infection.

Anticoagulant-Induced Skin Necrosis

Anticoagulant-induced skin necrosis is a term that includes heparin-induced thrombocytopenia (HIT) and warfarininduced skin necrosis (WISN).

Clinical Manifestations

Both HIT and WISN present with retiform purpura secondary to occlusion of cutaneous vessels. Necrosis may develop centrally secondary to tissue ischemia. HIT is most commonly found at sites of medication injection and typically develops within 5–10 days of starting heparin or within



Fig. 34.12 Heparin-induced thrombocytopenia: retiform purpura with central necrosis at the site of heparin injection on the abdomen

24 h in patients with recent exposure to heparin (Fig. 34.12). WISN typically develops 3–5 days after beginning warfarin, often on fatty sites, such as breasts, thighs, buttocks, and hips, and is preceded by pain.

Pathophysiology/Etiology

In HIT, circulating antibodies develop that simultaneously bind heparin and platelet factor 4 (PF4) causing platelet activation and subsequent aggregation leading to venous and arterial occlusion. For reasons that are not entirely clear, low-molecular-weight heparins are less likely to cause platelet activation than unfractionated heparins [54].

Warfarin-induced skin necrosis results from the temporary imbalance in pro- and anticoagulant factors upon initiation of warfarin. Warfarin functions by inhibiting vitamin K-dependent coagulation factors. Protein C, an anticoagulant, is more rapidly inhibited by warfarin than procoagulant factors II, VII, IX, and X, leading to a temporary prothrombotic state. Risk factors include obesity, perimenopausal age, viral infection, and underlying hypercoagulable state.

Diagnosis

In both HIT and WISN, skin biopsies reveal a pauciinflammatory thrombotic vasculopathy secondary to vessel occlusion. Subtle histologic variations in platelet thrombi (white clots) in HIT versus fibrin thrombi (red clots) in WISN can aid in differentiating the two histologically similar conditions [55]. In HIT, a rapid drop in platelets is typical resulting in absolute thrombocytopenia or a drop in platelets by at least 50%. The diagnosis of HIT can be confirmed with HIT antibody testing including immunoassays and functional assays. WISN is typically a clinical diagnosis that requires a compatible histology and history of recent initiation or reintroduction of warfarin. Patients may also have a history of an underlying hypercoagulable state or a recent infection. Protein C and S analyses are not sensitive or specific markers [56].

Treatment

Treatment of HIT consists of immediate discontinuation of heparin and supplementation with an alternative anticoagulant, such as fondaparinux, danaparoid, apixaban, rivaroxaban, or argatroban. Warfarin should be avoided initially but may be used once the patient has stabilized and platelet counts have recovered. Treatment of WISN involves immediate discontinuation of warfarin and administration of vitamin K and infusion of heparin at therapeutic doses. Fresh frozen plasma and protein C concentrate have been used to restore protein C levels and may be considered in the setting of life-threatening coagulation [56, 57]. It is recommended to bridge initiation of warfarin with heparin to avoid this phenomenon.

DIC/Purpura Fulminans

Infection, trauma, and malignancy among other insults can lead to imbalances in the coagulation system causing disseminated intravascular coagulation (DIC) with resultant simultaneous hemorrhage and thrombosis. If clotting is severe, purpura fulminans with acute, widespread retiform purpura and gangrene of the skin can develop.

Clinical Manifestations

DIC presents with skin findings indicative of simultaneous bleeding and thrombosis including petechiae, ecchymoses, and mucosal bleeding as well as lesions of retiform purpura. Purpura fulminans presents with rapidly progressive, wide-spread retiform purpura, hemorrhagic bulla, and symmetrical gangrene especially acrally (Fig. 34.13). Patients are acutely ill often with fever, shock, and evidence of multior-gan system involvement [58].

Pathophysiology/Etiology

Disseminated intravascular coagulation (DIC) and purpura fulminans are conditions in which systemic activation of coagulation leads to widespread clotting, particularly in small- and medium-sized vessels. Excessive clotting then leads to a consumptive coagulopathy where clotting factors cannot be generated as quickly as they are consumed and a bleeding diathesis ensues.

Diagnosis

Cardinal lab findings in DIC and purpura fulminans are consistent, independent of cause, and consist of thrombocytopenia, reduced plasma fibrinogen concentrations, and increased fibrin and fibrin products, with prolonged clotting times. Skin biopsy may aid in diagnosis and shows a thrombotic vasculopathy [58].



Fig. 34.13 Purpura fulminans: retiform purpura, blistering, and distal necrosis/gangrene of the hand

Treatment

Treatment of DIC includes treatment of the underlying cause and aggressive management of hemodynamic stability. If serious bleeding is present, platelet transfusions and administration of fresh frozen plasma or cryoprecipitate may be indicated. The use of heparin is typically limited to patients with chronic, compensated DIC that have predominantly thrombotic manifestations. Protein C concentrate may be considered in cases of DIC secondary to hereditary or acquired protein C deficiency or in cases of purpura fulminans. Adjunctive hyperbaric oxygen and surgical debridement with skin grafting may prove beneficial in patients with extensive skin necrosis and gangrene [59].

References

- Fiszenson-Albala F, Auzerie V, Mahe E, Farinotti R, Durand-Stocco C, Crickx B, et al. A 6-month prospective survey of cutaneous drug reactions in a hospital setting. Br J Dermatol. 2003;149:1018–22.
- Criado PR, Avancini J, Santi CG, Medrado AT, Rodrigues CE, de Carvalho JF. Drug reaction with eosinophilia and systemic symptoms (DRESS): a complex interaction of drugs, viruses, and immune system. Isr Med Assoc J. 2012;14:577–82.
- Naidoo J, Schindler K, Querfeld C, Busam K, Cunningham J, Page DB, et al. Autoimmune bullous skin disorders with immune checkpoint inhibitors targeting pd-1 and PD-L1. Cancer Immunol Res. 2016;4(5):383–9.
- Phillips GS, Wu J, Hellmann MD, Postow MA, Rizvi NA, Freites-Martinez A, et al. Treatment outcomes of immune-related cutaneous adverse events. J Clin Oncol. 2019;37(30):2746–58.
- National Cancer Institute. Division of Cancer Treatment & Diagnosis. Cancer Therapy Evaluation Program. Common Terminology Criteria for Adverse Events (CTCAE) v. 5.0. Updated

21 Sep 2020. Accessed 21 Sep 2020. https://ctep.cancer.gov/protocolDevelopment/electronic_applications/ctc.htm#ctc_50.

- Ahmed AR, Shetty S, Kaveri S, Spigelman ZS. Treatment of recalcitrant bullous pemphigoid (BP) with a novel protocol: a retrospective study with a 6-year follow-up. J Am Acad Dermatol. 2016;74(4):700–708 e703.
- Galván Casas C, Català A, Carretero Hernández G, Rodríguez-Jiménez P, Fernández-Nieto D, Rodríguez-Villa Lario A, et al. Classification of the cutaneous manifestations of COVID-19: a rapid prospective nationwide consensus study in Spain with 375 cases. Br J Dermatol. 2020;183(1):71–7.
- Blazar B, White ES, Couriel D. Understanding chronic GVHD from different angles. Biol Blood Marrow Transplant. 2012;18(1 Suppl):S184–8.
- Harris AC, Ferrara JL, Levine JE. Advances in predicting acute GVHD. Br J Haematol. 2013;160(3):288–302.
- Hirschmann JV, Raugi GJ. Lower limb cellulitis and its mimics: part I. Lower limb cellulitis. J Am Acad Dermatol. 2012;67(2):163. e1–12.;quiz 75–6.
- Hook IIIEW, Hooton TM, Horton CA, Coyle MB, Ramsey PG, Turck M. Microbiologic evaluation of cutaneous cellulitis in adults. Arch Intern Med. 1986;146(2):295–7.
- Bolognia JL, Cooper DL, Glusac EJ. Toxic erythema of chemotherapy: a useful clinical term. J Am Acad Dermatol. 2008;59(3):524–9.
- Canueto J, Meseguer-Yebra C, Roman-Curto C, Santos-Briz A, Fernandez-Lopez E, Fraile C, et al. Leukemic vasculitis: a rare pattern of leukemia cutis. J Cutan Pathol. 2011;38(4):360–4.
- Cho-Vega JH, Medeiros LJ, Prieto VG, Vega F. Leukemia cutis. Am J Clin Pathol. 2008;129(1):130–42.
- Bernstein JA, Lang DM, Khan DA, Craig T, Dreyfus D, Hsieh F, et al. The diagnosis and management of acute and chronic urticaria: 2014 update. J Allergy Clin Immunol. 2014;133(5):1270–7.
- Bhardwaj N, Craig TJ. Treatment of hereditary angioedema: a review (CME). Transfusion. 2014;54(11):2989–96; quiz 2988.
- Shands KN, Schmid GP, Dan BB, Blum D, Guidotti RJ, Hargrett NT, et al. Toxic-shock syndrome in menstruating women: association with tampon use and *Staphylococcus aureus* and clinical features in 52 cases. N Engl J Med. 1980;303(25):1436–42.
- Luca-Harari B, Ekelund K, van der Linden M, Staum-Kaltoft M, Hammerum AM, Jasir A. Clinical and epidemiological aspects of invasive *Streptococcus pyogenes* infections in Denmark during 2003 and 2004. J Clin Microbiol. 2008;46(1):79–86.
- Zimbelman J, Palmer A, Todd J. Improved outcome of clindamycin compared with beta-lactam antibiotic treatment for invasive *Streptococcus pyogenes* infection. Pediatr Infect Dis J. 1999;18(12):1096–100.
- Todd JK, Ressman M, Caston SA, Todd BH, Wiesenthal AM. Corticosteroid therapy for patients with toxic shock syndrome. JAMA. 1984;252(24):3399–402.
- Norrby-Teglund A, Ihendyane N, Darenberg J. Intravenous immunoglobulin adjunctive therapy in sepsis, with special emphasis on severe invasive group A streptococcal infections. Scand J Infect Dis. 2003;35(9):683–9.
- Wieselthier JS, Koh HK. Sézary syndrome: diagnosis, prognosis, and critical review of treatment options. J Am Acad Dermatol. 1990;22(3):381–401.
- 23. Siegel J, Totonchy M, Damsky W, Berk-Krauss J, Castiglione F Jr, Sznol M, et al. Bullous disorders associated with anti-PD-1 and anti-PD-L1 therapy: a retrospective analysis evaluating the clinical and histopathologic features, frequency, and impact on cancer therapy. J Am Acad Dermatol. 2018;79(6):1081–8.
- 24. Nelson CA, Singer S, Chen T, Puleo AE, Lian CG, Wei EX, et al. Bullous pemphigoid after anti-PD-1 therapy: a retrospective casecontrol study evaluating impact on tumor response and survival outcomes. J Am Acad Dermatol. 2020:S0190-9622(20)30048-7.
- 25. Nalamachu S, Morley-Forster P. Diagnosing and managing postherpetic neuralgia. Drugs Aging. 2012;29(11):863–9.
- Glesby MJ, Moore RD, Chaisson RE. Clinical spectrum of herpes zoster in adults infected with human immunodeficiency virus. Clin Infect Dis. 1995;21(2):370–5.
- 27. Schmutzhard J, Merete Riedel H, Zweygberg Wirgart B, Grillner L. Detection of herpes simplex virus type 1, herpes simplex virus type 2 and varicella-zoster virus in skin lesions. Comparison of real-time PCR, nested PCR and virus isolation. J Clin Virol. 2004;29(2):120.
- Chen CY, Huang SY, Tsay W, Yao M, Tang JL, Ko BS, et al. Clinical characteristics of candidaemia in adults with haematological malignancy, and antimicrobial susceptibilities of the isolates at a medical centre in Taiwan, 2001–2010. Int J Antimicrob Agents. 2012;40(6):533–8.
- Villada G, Roujeau JC, Cordonnier C, Bagot M, Kuentz M, Wechsler J, Vernant JP. Toxic epidermal necrolysis after bone marrow transplantation: study of nine cases. J Am Acad Dermatol. 1990;23(5 Pt 1):870–5.
- Coleman EL, Olamiju B, Leventhal JS. The life-threatening eruptions of immune checkpoint inhibitor therapy. Clin Dermatol. 2020;38(1):94–104.
- Chen CB, Wu MY, Ng CY, Lu CW, Wu J, Kao PH, et al. Severe cutaneous adverse reactions induced by targeted anticancer therapies and immunotherapies. Cancer Manag Res. 2018;10:1259–73.
- 32. Assier H, Bastuji-Garin S, Revuz J, Roujeau JC. Erythema multiforme with mucous membrane involvement and Stevens-Johnson syndrome are clinically different disorders with distinct causes. Arch Dermatol. 1995;131(5):539–43.
- Bastuji-Garin S, Rzany B, Stern RS, Shear NH, Naldi L, Roujeau JC. Clinical classification of cases of toxic epidermal necrolysis, Stevens-Johnson syndrome, and erythema multiforme. Arch Dermatol. 1993;129(1):92–6.
- Hsu DY, Brieva J, Silverberg NB, Silverberg JI. Morbidity and mortality of Stevens-Johnson syndrome and toxic epidermal necrolysis in United States adults. J Invest Dermatol. 2016;136(7):1387–97.
- 35. Micheletti RG, Chiesa-Fuxench Z, Noe MH, Stephen S, Aleshin M, Agarwal A, et al. Stevens-Johnson syndrome/toxic epidermal necrolysis: a multicenter retrospective study of 377 adult patients from the United States. J Invest Dermatol. 2018;138(11):2315–21. Erratum in: J Invest Dermatol. 2019;139(2):495–6.
- Rosen AC, Balagula Y, Raisch DW, Garg V, Nardone B, Larsen N, et al. Life-threatening dermatologic adverse events in oncology. Anti-Cancer Drugs. 2014;25(2):225–34.
- Maloney NJ, Ravi V, Cheng K, Bach DQ, Worswick S. Stevens-Johnson syndrome and toxic epidermal necrolysis-like reactions to checkpoint inhibitors: a systematic review. Int J Dermatol. 2020;59(6):e183–8.
- Palmieri TL, Greenhalgh DG, Saffle JR, Spence RJ, Peck MD, Jeng JC, et al. A multicenter review of toxic epidermal necrolysis treated in U.S. burn centers at the end of the twentieth century. J Burn Care Rehabil. 2002;23(2):87–96.
- Ng QX, De Deyn MLZQ, Venkatanarayanan N, Ho CYX, Yeo WS. A meta-analysis of cyclosporine treatment for Stevens-Johnson syndrome/toxic epidermal necrolysis. J Inflamm Res. 2018;11:135–42.
- 40. Wang CW, Yang LY, Chen CB, Ho HC, Hung SI, Yang CH, The Taiwan Severe Cutaneous Adverse Reaction (TSCAR) Consortium, et al. Randomized, controlled trial of TNF-α antagonist in CTLmediated severe cutaneous adverse reactions. J Clin Invest. 2018;128(3):985–96.
- 41. Zimmermann S, Sekula P, Venhoff M, Motschall E, Knaus J, Schumacher M, Mockenhaupt M. Systemic immunomodulating therapies for Stevens-Johnson syndrome and toxic epidermal necrolysis: a systematic review and meta-analysis. JAMA Dermatol. 2017;153(6):514–22.

- Kardaun SH, Jonkman MF. Dexamethasone pulse therapy for Stevens-Johnson syndrome/toxic epidermal necrolysis. Acta Derm Venereol. 2007;87(2):144–8.
- Halebian PH, Corder VJ, Madden MR, Finklestein JL, Shires GT. Improved burn center survival of patients with toxic epidermal necrolysis managed without corticosteroids. Ann Surg. 1986;204(5):503–12.
- 44. Huang YC, Li YC, Chen TJ. The efficacy of intravenous immunoglobulin for the treatment of toxic epidermal necrolysis: a systematic review and meta-analysis. Br J Dermatol. 2012;167(2):424–32.
- 45. Schneck J, Fagot JP, Sekula P, Sassolas B, Roujeau JC, Mockenhaupt M. Effects of treatments on the mortality of Stevens-Johnson syndrome and toxic epidermal necrolysis: a retrospective study on patients included in the prospective EuroSCAR study. J Am Acad Dermatol. 2008;58(1):33–40.
- Zimmermann J, Bahmer F, Rose C, Zillikens D, Schmidt E. Clinical and immunopathological spectrum of paraneoplastic pemphigus. J Dtsch Dermatol Ges. 2010;8(8):598–606.
- Wang L, Bu D, Yang Y, Chen X, Zhu X. Castleman's tumours and production of autoantibody in paraneoplastic pemphigus. Lancet. 2004;363(9408):525–31.
- 48. Hertl M, Zillikens D, Borradori L, Bruckner-Tuderman L, Burckhard H, Eming R, et al. Recommendations for the use of rituximab (anti-CD20 antibody) in the treatment of autoimmune bullous skin diseases. J Dtsch Dermatol Ges. 2008;6(5):366–73.
- Zhu X, Zhang B. Paraneoplastic pemphigus. J Dermatol. 2007;34(8):503–11.
- Campsall PA, Laupland KB, Niven DJ. Severe meningococcal infection: a review of epidemiology, diagnosis, and management. Crit Care Clin. 2013;29(3):393–409.
- 51. Cohn AC, MacNeil JR, Clark TA, Ortega-Sanchez IR, Briere EZ, Meissner HC, et al. Centers for Disease Control and Prevention (CDC). Prevention and control of meningococcal disease: recommendations of the advisory committee on immunization practices (ACIP). MMWR Recomm Rep. 2013;62(RR-2):1–28.
- Quatresooz P, Piérard-Franchimont C, Arrese JE, Piérard GE. Clinicopathologic presentations of dermatomycoses in cancer patients. J Eur Acad Dermatol Venereol. 2008;22(8):907–17.
- Paramythiotou E, Frantzeskaki F, Flevari A, Armaganidis A, Dimopoulos G. Invasive fungal infections in the ICU: how to approach, how to treat. Molecules. 2014;19(1):1085–119.
- Burns T, Breathnach S, Cox N, Griffiths C. Rook's textbook of dermatology. 8th ed. Oxford, UK: John Wiley-Blackwell; 2010.
- Nazarian RM, Van Cott EM, Zembowicz A, Duncan LM. Warfarininduced skin necrosis. J Am Acad Dermatol. 2009;61(2):325–32.
- Kakagia DD, Papanas N, Karadimas E, Polychronidis A. Warfarininduced skin necrosis. Ann Dermatol. 2014;26(1):96–8.
- Stewart A. Warfarin-induced skin necrosis treated with protein C concentrate (human). Am J Health Syst Pharm. 2010;67(11):901–4.
- Davis MD, Dy KM, Nelson S. Presentation and outcome of purpura fulminans associated with peripheral gangrene in 12 patients at Mayo Clinic. J Am Acad Dermatol. 2007;57(6):944–56.
- 59. Cooper JS, Allinson P, Keim L, Sisson J, Schuller D, Sippel J, Kovaleski DH. Hyperbaric oxygen: a useful adjunct for purpura fulminans: case report and review of the literature. Undersea Hyperb Med. 2014;41(1):51–7.
- Wharton M, Chorba TL, Vogt RL, Morse DL, Buehler JW. Case definitions for public health surveillance. MMWR Recomm Rep. 1990;39(RR-13):1–43.
- Centers for Disease Control and Prevention. Case definitions for infectious conditions under public health surveillance. MMWR Recomm Rep. 1997;46(RR-10):1–55.
- The Working Group on Severe Streptococcal Infections. Defining the group A streptococcal toxic shock syndrome. Rationale and consensus definition. JAMA. 1993;269(3):390–1.

Pediatrics

Kathleen M. Stephanos and Lindsey Picard

35

Introduction

Oncologic emergencies in pediatrics can be subdivided into presentation of disease and treatment of disease-associated emergencies. Each of these areas poses distinct risks and complications for the patient. This chapter will attempt to cover a multitude of different cancerous processes as well as their presentations. While many complications are similar to those in adults, the pathophysiology and resulting types of cancer can result in different treatment plans and prognostics. Many therapies are similar to those used in adults, with known adverse reactions, but their impact on the pediatric patient may result in more severe or long-term effects, a scenario which may become more common as cancer survivorship continues to improve.

Case Study

A 7-year-old female presents with 1 month of intermittent fevers, body aches, and left forearm pain. She has been seen by several providers since the onset of symptoms and diagnosed with viral illnesses. On the day of presentation, her parents described her as less alert throughout the day, so they brought her to the emergency department (ED). She has a temperature of 38.4, a heart rate of 130, a blood pressure of 85/50, a respiratory rate of 30, and an oxygen saturation of 100% on room air. On exam, she appears listless, with dry mucous membranes, her cardiopulmonary exam is normal other than the aforementioned vital signs, and she has diffuse lymphadenopathy and hepatosplenomegaly. What are the concerns for this patient? What is the differential diagnosis? What is the initial management? The child in the case presents initially with symptoms suggestive of acute leukemia. Symptoms supportive of this are her hepatosplenomegaly, intermittent fevers, and changes in mental status. Acute mental status changes may signify that this patient is experiencing CNS involvement of her illness or that she has hyperleukocytosis causing hyperviscosity symptoms. It is important to recognize this as a pediatric emergency that will require admission to a pediatric hospital with oncologic coverage. Unlike adults, who (if not in a blast crisis or having other requirements for admission) can often be managed as an outpatient, children are often admitted for initial oncologic evaluation.

Two months later, she returns with 1 hour of fever. She has been on chemotherapy since her primary presentation. She was tolerating her treatment well, other than occasional nausea and vomiting, until the day of presentation when she developed abdominal pain and has not been able to eat or drink. When the fever occurred, she was instructed by her oncologist to come directly to the ED. On arrival, her temperature is 39.2, heart rate 120, blood pressure 90/55, and respiratory rate 22, and she has an oxygen saturation of 99% on room air. On exam she appears uncomfortable and has red cracked lips and a dry tongue. Her cardiopulmonary exam is notable only for the aforementioned vital signs. On abdominal exam, her hepatosplenomegaly has resolved, but she now has right lower quadrant tenderness with rebound and guarding. What is on the differential for this patient? What is the management?

On her return visit, there are many factors to consider; however, the most pressing are recognition of the possibility of both neutropenic fever and typhlitis, both of which can be rapidly fatal if not identified early. This patient requires prompt recognition and should not be in a waiting room for any duration of time. Other considerations include possible mucositis with dehydration or bowel perforation. She may also have appendicitis or even intussusception based on her pain. Her immunosuppressed state may have delayed the common symptoms of these, and her medications may have mimicked some of these symptoms as well causing family

K. M. Stephanos $(\boxtimes) \cdot L$. Picard

Department of Emergency Medicine, Strong Memorial Hospital, University of Rochester Medical Center, Rochester, NY, USA e-mail: kathleen_stephanos@urmc.rochester.edu

K. H. Todd et al. (eds.), Oncologic Emergency Medicine, https://doi.org/10.1007/978-3-030-67123-5_35

and providers to attribute her recent nausea and vomiting due to medication effects rather than early signs of disease. Based on her presentation, intravenous fluid should be started, with antibiotics and imaging for better assessment of her abdomen.

History and Background

Overall incidence of pediatric cancer has not changed significantly over the past several decades; however, there has been one drastic change: on average, more than four out of every five children diagnosed with cancer can be expected to survive. A particular challenge of pediatric cancer is the small number of cases, resulting in limited testing and evaluation of new treatment options [1]. Pediatric cancers are managed several ways including multidisciplinary therapeutic strategies such as surgical intervention, cytotoxic chemotherapy, radiation therapy, and autologous as well as allogeneic bone marrow or stem cell transplants. More recent studies have begun to evaluate the efficacy of monoclonal antibody therapy, though this is not a commonly used therapy at the time of this publication [2, 3]. The treatment options are cancer specific, but nearly all pediatric cancers acquire some form of significant immunosuppression.

Anatomy

As with all oncologic processes, pediatric cancers are due to an unregulated overgrowth of mutated cells. The most common sites for this in children are bone marrow and central nervous system—specifically intracranial [4]. Therapy, therefore, targets these cells specifically.

Pathophysiology

New Onset

There are a variety of different pathophysiologic pathways depending on the type of pediatric cancer. In many pediatric cancers, the different hematopoietic cell lines are affected, which in turn leads to anemia, leukocytosis, and thrombocytopenia. A normocytic, normochromic anemia results from bone marrow infiltration (seen in leukemia), as an inflammatory process from cytokine release resulting in decreased production or from direct blood loss [5]. Similar effects are seen on platelets. Hyperleukocytosis can be seen in different pediatric cancers, defined as white blood cell count above 100,000 cells/mm³. It can result in leukostasis leading to pulmonary and CNS complications [5].

Tumor lysis is another feature of cancer pathophysiology. It can occur as a result of high tumor burden and cell turnover or from initial treatment. It leads to the release of intracellular components into the blood, including DNA and electrolytes. Multiple electrolyte abnormalities are seen in tumor lysis syndrome. Renal failure, cardiac dysrhythmias, and seizures can result [6].

Undergoing Therapy

Treatment of disease targets the rapidly dividing mutated cells in order to prevent their further growth. This has known consequences on the surrounding cells, as well as other highly replicative cells within the human body including hair, blood, and mucosal, among others. Many chemotherapeutic agents result in severe, but often anticipated, adverse side effects. Common sequelae include alopecia, mucositis, vomiting, and diarrhea, accompanied by malnutrition. Often, these complications are managed in an outpatient setting. For example, prevention of emesis with 5-HT3 antagonists is common practice [7]. Pediatric infusion centers can administer intravenous fluids and prescribe antiemetics. For more severe emesis episodes, a child may require admission or presentation to an emergency department for management of hydration [8]. Immediately following initiation of chemotherapy, a patient may develop tumor lysis syndrome, as mentioned above. In known oncologic patients, this is typically seen on initiation of therapy only, during which the patient is already admitted and being monitored and would be an unanticipated finding in an ED. Beyond these anticipated outcomes, many of the current chemotherapeutic agents have severe impacts on organs throughout the body, some which are temporary and others that are more permanent [8].

Epidemiology

Pediatric cancers are the second most common cause of death due to disease in children and adolescents, behind only accidents. In 2018, there were over 10,000 new cancer diagnoses in children under age 14. Leukemias are the most common cancer diagnosis, followed by central nervous system (CNS) malignancies in young children. Adolescents have a different distribution; CNS cancers and lymphomas are the most common forms found in this age group [4].

Health Economics

A pediatric cancer admission costs roughly \$40,000, which is five times more expensive than an average general pediatric hospital admission [9]. In addition to the medical costs from hospital admissions, cancer treatments, and outpatient follow-ups, many families experience increased financial burdens as parents commonly experience employment disruptions [9]. A recent study demonstrated that advanced childhood cancer takes a disproportionate toll on family finances when compared to other complex chronic pediatric patients [10]. In one study, nearly 45% of families of pediatric patients had an identifiable socioeconomic or resource limitation hardship [11]. All of these contextual factors are important for the ED provider to appreciate for this patient cohort.

Presentation/Diagnosis

New Onset

Many initial cancer presentations in the pediatric population are nonspecific. Typically they present with vague symptoms, such as fatigue, headache, pallor, weight loss, unexplained bone pain, or fevers [12]. In young children, any unexplained weight loss is concerning, since children normally will follow a progressive growth curve. Even those children attempting to become less obese will target limiting weight gain rather than overt weight loss unless under specific guidance or weight loss planning. There are some more specific presentations to be aware of; for example, new-onset wheezing, particularly in school-aged children, can indicate a mediastinal mass [13]. Altered mental status, focal neurologic deficits, and seizures without return to baseline could be secondary to an intracranial malignancy [12]. This can also be due to spread of a hematologic malignancy or hyperleukocytosis. Metastatic brain lesions are an exceptionally uncommon presentation of disease in children, which is notably different than in adults. Unexplained bruising, bleeding, or petechiae are suggestive of thrombocytopenia, which could be from a malignant process. Intermittent abdominal pain, vomiting, constipation, or bloody stools could indicate intra-abdominal cancer pathology. Neuroblastoma may cause catecholamine-induced hypertension with cardiomyopathy or parasympathetic-driven refractory diarrhea, a-traumatic "raccoon eyes," or abnormal movements [14]. Superior vena cava syndrome may occur in children, though is less common than in adults, and can be due to thrombus or mass compression. This presents with cough, dyspnea, and swelling or color changes to the head and neck [6]. Tumor lysis may be found at presentation due to solid organ tumors. This is due to tumor involution and breakdown due to the tumor overgrowing its vascular supply, at which time tumor cells may rapidly break down, releasing electrolytes into the bloodstream. As the kidneys attempt to filter these lysed cells, renal function decreases, and the patient may suffer acute renal failure. Likewise, the released electrolytes, most specifically potassium and phosphorus, can have significant cardiac and other systemic effects [6] (Table 35.1).

Cancer	Symptoms
Leukemia,	Fever, bone pain, bruising, bleeding,
lymphoma	lymphadenopathy
Mediastinal masses	Wheezing, cough
Neuroblastoma	Diarrhea, hypertension, constipation, abdominal pain, periorbital ecchymosis, cardiomyopathy
Abdominal	Constipation, abdominal pain, vomiting, bloody
masses	stools
Intracranial	Altered mental status, seizures, neurologic
masses	deficits, headache, vomiting

Undergoing Therapy

There are a wide variety of complications due to cancer treatment. Surgical interventions may cause typical postsurgical issues including bleeding, pain, or infection. Medical treatments, however, may have systemic effects posing risks to nearly every organ system.

Blood dyscrasias are one of the most common side effects of chemotherapeutic agents and are typically the result of bone marrow suppression. These include leukopenia, thrombocytopenia, and anemia. Specifically, neutropenia may be severe resulting in increased risks for viral, bacterial, and fungal infections [15]. In these patients, fever can be the first sign of a severe systemic infection, which can, in turn, result in rapid and severe decompensation, particularly if not recognized and managed quickly [15]. Due to immunosuppression, patients may not show typical early signs of infection, including pain. Additionally, symptoms of infection may be mistaken for other common medication side effects, such as fatigue or pallor. Neutropenic fever comprised about 19% of post-diagnosis ED presentations in one study, and infections (or signs and symptoms of infection) accounted for over 40% of all visits [8].

Anemia may be the result of increased bleeding or bone marrow suppression [16]. Bleeding is commonly due to thrombocytopenia, though can also be due to coagulopathic side effects of medications [5]. Most oncologic patients have treatment-specific target transfusion criteria, though it is important to recognize that the need for frequent transfusions poses a risk for iron overload [12, 17].

While there is much focus on the bone marrow suppressive implications for the hematopoietic system, there is also a risk of hypercoagulation. Overall pulmonary embolism is very rare in children, but it has an incidence of 2% on routine CT (not specifically looking for pulmonary embolism) in oncologic patients. This hypercoagulopathic side effect also increases the risk of stroke, superior vena cava syndrome, as well as other ischemic diseases [12]. Thrombosis may also be related to indwelling central lines [16]. Hemoptysis may develop due to easy bleeding as a result of low platelets, pulmonary embolism, pulmonary edema, or, very commonly, infection. Diffuse alveolar hemorrhage may also occur and is associated with hematopoietic stem cell transplant [14]. This is also found in 40% of pediatric cancer autopsies and is the most frequently missed diagnosis in pediatric cancer deaths [18].

Cardiovascular compromise secondary to treatment may present early or be delayed. Pericardial effusion and cardiac tamponade may present, similar to adults, with chest pain, respiratory symptoms (such as cough or dyspnea), tachycardia, and occasionally pulsus paradoxus [19]. Specific therapies (e.g., anthracyclines) may also cause arrhythmias as well as heart failure resulting in long-term cardiac effects [20].

In the gastrointestinal tract, one of the most frequent side effects of cancer treatment is mucositis [16]. This is due to the breakdown of mucosal tissue throughout the intestinal tract, which is painful and increases the risk of bleeding, infection, and dehydration [21]. Bowel perforation may be found on presentation or due to treatment of disease. Steroids as well as chemotherapeutic agents increase the propensity for perforation due to structural effects on the bowel [21]. Patients may present with minimal initial symptoms due to immunosuppression. One of the most severe intra-abdominal complications of chemotherapy is neutropenic enterocolitis (also called typhlitis or ileocecal syndrome). This lifethreatening disease process typically presents in severely neutropenic patients and is often associated with leukemia or lymphoma. Patients report abdominal pain, which is commonly located along the colon and often in the right lower quadrant. They may also have fever or diarrhea [22]. It is believed that this is due to chemotherapy causing friable mucosa vulnerable to infectious infiltration [22].

Pancreatitis can be caused by a variety of therapeutic agents including corticosteroids, trimethoprimsulfamethoxazole (to prevent immunosuppression-related pneumocystis pneumonia), asparaginase, mercaptopurine, and cytarabine [14].

Renal system injuries are categorized as subacute and acute. Subacute injury is often due to treatment modalities and is less likely to present in extremis to the ED, as these patients are closely monitored for blood dyscrasias and electrolyte abnormalities during treatment [23]. Acutely, renal injury may occur due to tumor lysis syndrome. Hemorrhagic cystitis is a common complication of chemotherapy, specifically cyclophosphamide [24].

Graft-versus-host disease can cause a myriad of symptoms throughout the entire body. This typically occurs following allogeneic bone marrow transplantation. Symptoms may include skin findings, dyspnea, myalgias, fatigue, fever, liver disease, and other gastrointestinal effects [14].

Electrolyte abnormalities are also possible with a variety of chemotherapeutic agents (Table 35.2) [25–38]. Specifically

 Table 35.2
 A review of common severe side effects for patients on various chemotherapeutic agents

Medication	Life-threatening complications	
Asparaginase	Hemorrhage, coagulopathy (thrombosis,	
	emboli) hypotension, pancreatitis,	
	bronchospasm, decreased blood counts	
Busulfan, myleran	Thrombosis, electrolyte abnormalities, severe	
	fatigue, renal damage, decreased blood counts	
Carboplatin	Renal damage, decreased blood counts,	
	electrolyte abnormalities, bone marrow	
	suppression	
Cisplatin	Decreased blood counts, allergy, renal damage,	
	electrolyte abnormalities, peripheral	
	neuropathy	
Cyclophosphamide	Bladder damage (hemorrhagic cystitis), oral	
~	sores, infertility	
Cytarabine	Decreased blood counts, fever, flu-like	
	symptoms, aseptic meningitis, neurotoxicity,	
	intestinal necrosis, toxic megacolon,	
D 111	pancreatitis, rnabdomyolysis	
Daunorubicin,	Decreased blood counts, heart failure,	
doxorubicin	arrhythmias (note also changes in the color of	
Et	sweat and urine)	
Etoposide	Hypotension, peripheral neuropathy	
Hydroxyurea	Decreased blood cell counts (macrocytosis,	
	neutropenia), infection, severe central nervous	
Managatanyaina	Degreesed blood counts, four remonstitie	
Mercaptopurme	renal toxicity mucositis, pulmonary fibrosis	
Mathotrovoto	Degrapsed blood counts, disbates, intesting	
Methotrexate	perforation dizziness headache focal	
	neurologic findings rashes infertility severe	
	renal disease intestinal bleeding mucositis	
	tumor lysis syndrome, obstructive and	
	restrictive lung diseases	
Thioguanine	Decreased blood counts, portal hypertension.	
Thioguanite	liver failure	
Thiotepa	Intracranial hemorrhage, seizure, mucositis,	
Interpr	decreased blood counts, infertility, asthma	
Topotecan	Myalgias, decreased blood counts, dyspnea	
Vincristine	Nerve injuries (central and peripheral)	
	ischemic heart disease, decreased blood	
	counts, intestinal perforation and necrosis.	
	decreased blood counts. Hepatic sinusoidal	
	obstruction	

This list excludes common symptoms that are unlike to present to an emergency department or require significant intervention—such as nausea and vomiting [25–38]

vincristine, cyclophosphamide, and cisplatin are associated with syndrome of inappropriate anti-diuretic hormone (SIADH), resulting in hyponatremia [16].

Evaluation

New Onset

Given the typical, vague initial presentation for pediatric cancers, there is no set ED protocol for evaluation of these patients. Work-up should be based on history and physical examination. Laboratory studies may include complete blood count with differential, electrolytes, renal function, lactate dehydrogenase, and uric acid. Imaging may be indicated, such as chest x-ray, head CT or MRI, abdominal ultrasound, or extremity imaging. Consultation with hematology/ oncologic colleagues will help guide additional laboratory and imaging studies.

When there is concern for tumor lysis syndrome, lab work should focus on electrolytes, as well as lactate dehydrogenase, uric acid, calcium, and phosphate [6]. Elevated potassium, phosphate, urea nitrogen, uric acid, and lactate dehydrogenase, accompanied by low calcium, are supportive of this diagnosis [6].

Undergoing Therapy

Each therapeutic tool has different, treatment-specific risks. Radiation therapy typically causes local inflammatory reactions. While nearly all chemotherapeutic medications can result in decreased blood counts due to bone marrow suppression, as well as nausea, vomiting, anorexia, or alopecia, some have more severe complications specifically associated with their use. (See Table 35.2 for highrisk complications associated with specific medications.) Asking the patient or family member to identify the medications may help the provider assess for the risk for specific diagnoses.

In general, patients with oncologic processes who are undergoing treatment require rapid ED assessment and prompt placement in a treatment room isolated from other patients and more specifically from patients with infectious symptoms [39]. These oncologic patients typically warrant blood work including complete blood counts with differential, allowing assessment for neutropenia, anemia, and thrombocytopenia. Absolute neutrophil count should be determined using the following formula: (percent neutrophils + percent bands) × white blood cells × 100 [40]. A neutrophil count below 1000 cells/mm³ is considered neutropenic, and below 500 cells/mm³, severely neutropenic [4]. A basic metabolic panel assesses hydration and electrolyte status. Finally, liver function testing and pancreatic tests, specifically lipase, are often warranted in patients with nonspecific infection symptoms and should be obtained in patients with any gastrointestinal symptoms [16]. The emergency physician should communicate with the patient's oncologist or a member of their team early in evaluation and management to ensure appropriate medications and treatments are given and risk factors are identified. Infectious evaluation of pediatric patients should include complete blood counts, cultures (including cultures from any central access points), urinalysis, and urine culture. If there is concern for neutropenia, urine should not be obtained via catheter due to the risk for introduction of infection. Concern for meningitis may require meningitic dosing of antibiotics and avoidance of lumbar puncture in a thrombocytopenic patient [15].

Diffuse alveolar hemorrhage may manifest as bilateral opacities on chest x-ray or ground-glass opacities on CT. Chest radiographs may reveal cardiogenic pulmonary edema and cardiomegaly [14]. For pulmonary embolism, CT angiography is the test of choice. MRA, while having some utility in adult patients, has not been assessed for pediatrics and therefore has an unclear role. Nuclear medicine or interventional radiology studies are rarely utilized in children [14].

If there is concern for pericardial effusion and/or tamponade, chest x-ray may reveal a water-bottle heart appearance; though this has some specificity, its sensitivity is poor [12]. Echocardiography is the test of choice to evaluate pericardial effusion.

Plain abdominal radiographs may find thickened bowel walls or air-fluid levels, suggesting graft-versus-host disease, typhlitis, or other intra-abdominal pathology. Ultrasound may also be utilized in pediatric patients to assess bowel wall thickness, with levels over 3 mm suggestive of thickening [14]. Ultrasound avoids radiation and allows more accurate measurement of bowel wall thickness than CT. Pneumatosis intestinalis indicating typhlitis may be seen on a screening x-ray, but CT should be performed as well, assuming the patient is stable enough for transport.

Patients with pancreatitis do not necessarily require emergent imaging; however, ultrasound is the evaluation of choice. Again, radiation is limited, and an enlarged or hypoechoic pancreas, including pseudocysts, may be seen. For patients in whom ultrasound is equivocal or unavailable, CT scan is an appropriate modality for assessment [14]. Of course, pediatric cancer patients may have typical diseases of childhood (e.g., appendicitis, intussusception) and should have standard of care performed for these diagnoses as well.

Management

New Onset

The management of new cancer diagnoses in the ED is most frequently supportive. As mentioned above, consultation with pediatric hematology and oncologic is important. Their expertise will guide the emergency physician through the necessary primary evaluation. Initial presentations can include tumor lysis and hyperleukocytosis, both of which are treated initially with intravenous fluids. Allopurinol and rasburicase can be added as needed to treat tumor lysis [6]. Hyperleukocytosis may require leukapheresis and high volume intravenous fluids [12]. These additional medications should only be added when discussed with pediatric hematology and oncologic. Patients with mediastinal masses with concern for airway compression may require intubation. These patients also highlight the importance of imaging for new wheezing. Management should not include steroids as this can impact the tissue and limit the ability to make an accurate diagnosis on biopsy.

Neuroblastoma patients may require blood pressure management, when experiencing blood pressure elevations above the 95th percentile for age and height or if it is associated with end-organ dysfunction.

A key component to the initial management of a new cancer diagnosis is the discussion with the patient and their family. Discussing the diagnosis first with the parents is vital as it allows the parents to cope with this life-changing news. In addition, it allows for the parents and physician to consider the most appropriate way to tell the patient. The patient should be informed using age-appropriate terminology. A child life specialist may assist in this discussion with the patient and family when one is available. Ideally, children who are old enough to understand would be informed of the diagnosis shortly after parents are notified. There is evidence that delays in informing children can result in distrust of both parents and the medical system [41].

Undergoing Therapy

Patients presenting with isolated blood dyscrasias may require transfusion to meet the goals of their therapeutic plan. Blood should be irradiated and leukocyte reduced [17]. Additionally, cytomegalovirus (CMV) is less common in pediatric than adult patients, and CMV-negative blood products should be administered unless there is documentation that the patient has tested positive for CMV antibodies [5]. Transfusions in pediatrics are typically administered based on weight, 10 ml/kg up to 250–300 ml (the typical adult unit of blood), and can be expected to raise hemoglobin levels by 2–3 g/dL [16].

Coagulopathy resulting in increased clot burden may require anticoagulants. This must be done cautiously in order to avoid bleeding and worsening anemia [5].

Fever in the setting of known or possible neutropenia is considered an emergency, and these patients require prompt antibiotic therapy. While ampicillin-sulbactam and cefepime are often cited as the antibiotics of choice, therapy should be tailored to the patient's past sensitivity and resistance patterns when possible [15, 16]. If this information is not readily available to the emergency provider, the patient's oncologist should be consulted. In ill-appearing patients, the addition of vancomycin should be added to management early in treatment [15].

Pericardial effusion may require placement of a pericardial drain or surgical intervention with a pericardial window [19]. In cases of tamponade physiology with cardiovascular collapse, emergent pericardiocentesis may be indicated [29].

Pancreatitis is treated similarly in children as in adults, with initiation of intravenous fluids and bowel rest [42]. Intra-abdominal processes such as typhlitis or graft-versus-host disease require the initiation of antibiotics and surgical consultation [22].

Electrolyte repletion should be monitored closely and based on the severity of the imbalance [16].

Of note, the specific diagnosis of COVID-19 has not been shown to have significant morbidity and mortality in pediatric oncologic patients to date, though understanding of this disease and its impact on this patient population is actively changing [43].

Disposition/Follow-Up

New Onset

Unlike the adult population, new cancer diagnoses in pediatric patients typically require hospital admission. During the admission, thorough work-up of the new cancer will occur with discussions of management options. Depending on the cancer type, initiation of treatment may occur during this admission. Coordination of care is key during the admission to facilitate accurate diagnosis and treatment, as well as provide the necessary patient and family support.

Undergoing Therapy

The majority of pediatric patients who present to an ED while undergoing chemotherapy will require hospital admission [8]. Some exceptions include those who require simple transfusions or patients with concern for neutropenic fever who both appear well and are not found to have neutropenia. Disposition of all patients with treatment-associated complications should be in conjunction with the patient's primary oncologic team.

Prognosis/Treatment

Over 80% of children will pass the 5-year survivor mark. In a 2020 review by Horn et al., the overall mortality rate of pediatric patients with a cancer diagnosis was estimated to be 23% with 73% of deaths resulting from their primary cancer and 27% from a competing cause [44]. Deaths from competing causes were most closely associated with Hodgkin lymphoma and gonadal germ cell tumors, both highly treatable with current chemotherapeutic agents. In both of these cancers, as well as with osteosarcoma, the cause of death is often a secondary cancer that may be treatment-related [44].

Complications

With rapid advances in therapy, many more pediatric patients may survive into adulthood. Long-term sequelae of some treatments used in children are still being discovered. It is known that some patients have secondary treatment-related cancers, as well as long-term cardiac effects [44]. It is also important to note that pediatric cancer survivors have a higher than typical risk of suicide. This typically occurs in the third or fourth decade of life, indicating a need for longterm psychosocial support, and not just during active treatment [44]. It is important for emergency physicians to be aware of this and consider previous cancer as a risk factor when screening children and adults in the ED.

Common Pitfalls

New Onset

The most common pitfall of diagnosis is not considering an oncologic process in the differential of a vague complaint. A missed diagnosis can delay care or in severe cases be fatal. Additionally it is important avoid steroids in patients who present with new-onset wheezing without considering the possibility of a mediastinal mass.

Undergoing Therapy

The most common pitfall in patients receiving treatment is not recognizing the severe impact chemotherapeutic agents have on typical symptom presentations. The immunosuppressive therapies may also suppress typical systemic symptoms of disease.

Prevention/Upstream Drivers

Unlike many adult oncologic processes, there are very few pediatric cancers with specific, preventable causes. There is a significant increase in cancer among pediatric patients receiving organ transplants, primarily due to Epstein-Barr virus infection in conjunction with immunosuppression to prevent rejection [45]. Socioeconomics has not been linked to leukemic processes [46]. While some studies speculate that children born via artificial reproductive technologies have a higher risk of cancer, some recent studies refute this claim [47, 48]. Birth defects are associated with increased cancer risk, with some defects related to specific cancers [49]. Overall, genetic predisposition is believed to be the more common cause of disease than specific environmental exposures [50]. Prevention and recognition of secondary treatment-related complications should be enhanced with close monitoring by the patient's oncologist.

Future Needs/Vision

Pediatric oncologic is a rapidly changing field of medicine. Within the past 20 years, survival rates continue to improve. The rise of monoclonal antibody use in adults may soon be employed in pediatrics as well, with studies already being conducted with neuroblastoma [2, 3]. Emergency physicians should be aware that the complications of these medications in children are not well known; however, they may be similar to those in adults. Current studies suggest that the same weight-based dosing results in lower plasma levels in infants and children, which must be taken into consideration for future management [51].

Health Services/Resource Utilization

Studies show that, even with the same malignancy, pediatric patients, and specifically adolescent patients, have higher costs associated with treatment compared to adults. These increased costs are likely related to increased rates of hospitalization [52].

Policy/Legislative

Medicolegal

The most important component of ED management of patients with known oncologic processes is rapid evaluation. In these patients, as with every neutropenic fever patient, time is essential for infection control and hemorrhage management [15]. When available, notification prior to arrival with pre-ordering treatment for specific diagnoses, such as neutropenic fever, may improve time to administration of live-saving antibiotics.

Documentation/Quality Indicators

The main quality indicators in pediatric oncologic are focused on time to antibiotics in the setting of neutropenic fever. It is important to document the initial examination in detail as it may change rapidly, as well as repeat examinations that give a clear picture of any changes that occur.

Key Points/Pearls

Pattern recognition is key in pediatric cancer diagnosis. Staying vigilant is vital to avoid missing key diagnoses. Weight loss in children, unexplained bruising or bleeding, persistent and unexplained pain in back or extremities, newonset wheezing, and mental status changes are particularly pertinent to cancer diagnoses. The emergency provider should maintain a low threshold for detailed evaluation of the oncologic patient undergoing active therapy.

References

- Saletta F, Seng MS, Lau LMS. Advances in paediatric cancer treatment. Transl Pediatr. 2014;3(2):156–82.
- Kushner BH, Cheung IY, Modak S, Basu EM, Roberts SS, Cheung NK. Humanized 3F8 anti-GD2 monoclonal antibody dosing with granulocyte-macrophage colony-stimulating factor in patients with resistant neuroblastoma: a phase 1 clinical trial. JAMA Oncol. 2018;4(12):1729–35.
- 3. Ploessl C, Pan A, Maples KT, Lowe DK. Dinutuximab: an antigd2 monoclonal antibody for high-risk neuroblastoma. Ann Pharmacother. 2016;50(5):416–22.
- Siegel RL, Miller KD, Jemal A. Cancer statistics, 2018. CA Cancer J Clin. 2018;68(1):7–30.
- Mickley ME, Gutierrez C, Carney M. Oncologic and hematologic emergencies in children. In: Tintinalli JE, Stapczynski J, Ma O, et al., editors. Tintinalli's emergency medicine: a comprehensive study guide. New York: McGraw-Hill; 2016. p. 958–76.
- 6. Gupta A, Moore JA. Tumor lysis syndrome. JAMA. 2018;4(6):895.
- Patel P, Robinson PD, Thackray J, Flank J, Holdsworth MT, Gibson P, et al. Guideline for the prevention of acute chemotherapyinduced nausea and vomiting in pediatric cancer patients: a focused update. Pediatr Blood Cancer. 2017;64(10) https://doi.org/10.1002/ pbc.26542.
- Mueller EL, Hall M, Shah SS, August KJ, Radhi MA, Macy ML. Characteristics of children with cancer discharged or admitted from the emergency department. Pediatr Blood Cancer. 2016;63(5):853–8.
- Warner E, Kirchhoff A, Nam G, Fluchel M. Financial burden of pediatric cancer for patients and their families. J Oncol Pract. 2015;11(1):12–8.
- Bona K, Dussel V, Orellana L, Kang T, Geyer R, Feudtner C, Wolfe J. Economic impact of advanced pediatric cancer on families. J Pain Symptom Manag. 2014;47(3):594–603.
- Zheng DJ, Shyr D, Ma C, Muriel AC, Wolfe J, Bona K. Feasibility of systematic poverty screening in a pediatric oncology referral center. Pediatr Blood Cancer. 2018;65(12):e27380.
- Freedman JL, Rheingold SR. Management of oncologic emergencies. In: Lanzkowsky P, Lipton JM, Fish JD, editors. Lanzkowsky's manual of pediatric hematology and oncology. New York: Elsevier; 2016. p. 605–19.
- Lam JC, Chui CH, Jacobsen AS, Tan AM, Joseph VT. When is a mediastinal mass critical in a child? An analysis of 29 patients. Pediatr Surg Int. 2004;20(3):180–4.
- Handa A, Nozaki T, Makidono A, Okabe T, Morita Y, Fujita K, et al. Pediatric oncologic emergencies: clinical and imaging review for pediatricians. Pediatr Int. 2019;61(2):122–39.
- Lehrnbecher T, Robinson P, Fisher B, Alexander S, Ammann RA, Beauchemin M, et al. Guideline for the management of fever and neutropenia in children with cancer and hematopoietic stem-cell transplantation recipients: 2017 update. J Clin Oncol. 2017;35(18):2082–94.
- Prusakowski MK, Cannone D. Pediatric oncologic emergencies. Hematol Clin N Am. 2017;31(6):959–80.
- Bercovitz RS, Josephson CD. Thrombocytopenia and bleeding in pediatric oncology patients. Hematology Am Soc Hematol Educ Program. 2012;2012:499–505.
- Furuya ME, Rodríguez-Velasco A, Rodríguez-Zepeda MC, López-Aguilar E, Vargas MH, Sciandra-Rico M, et al. Unsuspected lung

pathology in autopsies of children with cancer. Rev Investig Clin. 2017;69(1):28–32.

- Kühn B, Peters J, Marx GR, Breitbart RE. Etiology, management, and outcome of pediatric pericardial effusions. Pediatr Cardiol. 2008;29(1):90–4.
- Bansal N, Amdani S, Lipshultz ER, Lipshultz SE. Chemotherapyinduced cardiotoxicity in children. Expert Opin Drug Metab Toxicol. 2017;13(8):817–32.
- Kuiken NS, Rings EH, Tissing WJ. Risk analysis, diagnosis and management of gastrointestinal mucositis in pediatric cancer patients. Crit Rev Oncol Hematol. 2015;94(1):87–97.
- Rodrigues FG, Dasilva G, Wexner SD. Neutropenic enterocolitis. World J Gastroenterol. 2017;23(1):42–7.
- Ruggiero A, Ferrara P, Attinà G, Rizzo D, Riccardi R. Renal toxicity and chemotherapy in children with cancer. Br J Clin Pharmacol. 2017;83(12):2605–14.
- Cyclophosphamide. In: Specific Lexicomp Online Database [database on the Internet]. Hudson, OH: Lexicomp Inc. 2020; cited 30 Apr 2020. Available from: http://online.lexi.com. Subscription required to view.
- Asparaginase. In: Specific Lexicomp Online Database [database on the Internet]. Hudson, OH: Lexicomp Inc. 2020; cited 30 Apr 2020. Available from: http://online.lexi.com. Subscription required to view.
- Busulfan. In: Specific Lexicomp Online Database [database on the Internet]. Hudson, OH: Lexicomp Inc. 2020; cited 30 Apr 2020. Available from: http://online.lexi.com. Subscription required to view.
- Carboplatin. In: Specific Lexicomp Online Database [database on the Internet]. Hudson, OH: Lexicomp Inc. 2020; cited 30 Apr 2020. Available from: http://online.lexi.com. Subscription required to view.
- Cisplatin. In: Specific Lexicomp Online Database [database on the Internet]. Hudson, OH: Lexicomp Inc. 2020; cited 30 Apr 2020. Available from: http://online.lexi.com. Subscription required to view.
- Cytarabine. In: Specific Lexicomp Online Database [database on the Internet]. Hudson, OH: Lexicomp Inc. 2020; cited 30 Apr 2020. Available from: http://online.lexi.com. Subscription required to view.
- Daunorubicin. In: Specific Lexicomp Online Database [database on the Internet]. Hudson, OH: Lexicomp Inc. 2020; cited 30 Apr 2020. Available from: http://online.lexi.com. Subscription required to view.
- Etoposide. In: Specific Lexicomp Online Database [database on the Internet]. Hudson, OH: Lexicomp Inc. 2020; cited 30 Apr 2020. Available from: http://online.lexi.com. Subscription required to view.
- Hydroxyurea. In: Specific Lexicomp Online Database [database on the Internet]. Hudson, OH: Lexicomp Inc. 2020; cited 30 Apr 2020. Available from: http://online.lexi.com. Subscription required to view.
- Mercaptopurine. In: Specific Lexicomp Online Database [database on the Internet]. Hudson, OH: Lexicomp Inc. 2020; cited 30 Apr 2020. Available from: http://online.lexi.com. Subscription required to view.
- Methotrexate. In: Specific Lexicomp Online Database [database on the Internet]. Hudson, OH: Lexicomp Inc. 2020; cited 30 Apr 2020. Available from: http://online.lexi.com. Subscription required to view.
- 35. Thioguanine. In: Specific Lexicomp Online Database [database on the Internet]. Hudson, OH: Lexicomp Inc. 2020; cited 30 Apr 2020. Available from: http://online.lexi.com. Subscription required to view.
- Thiotepa. In: Specific Lexicomp Online Database [database on the Internet]. Hudson, OH: Lexicomp Inc. 2020; cited 30 Apr 2020.

Available from: http://online.lexi.com. Subscription required to view.

- Topotecan. In: Specific Lexicomp Online Database [database on the Internet]. Hudson, OH: Lexicomp Inc. 2020; cited 30 Apr 2020. Available from: http://online.lexi.com. Subscription required to view.
- Vincristine. In: Specific Lexicomp Online Database [database on the Internet]. Hudson, OH: Lexicomp Inc. 2020; cited 30 Apr 2020. Available from: http://online.lexi.com. Subscription required to view.
- 39. Salstrom JL, Coughlin RL, Pool K, Bojan M, Mediavilla C, Schwent W, Rannie M, Law D, Finnerty M, Hilden J. Pediatric patients who receive antibiotics for fever and neutropenia in less than 60 min have decreased intensive care needs. Pediatr Blood Cancer. 2015;62(5):807–15.
- Al-Gwaiz LA, Babay HH. The diagnostic value of absolute neutrophil count, band count and morphologic changes of neutrophils in predicting bacterial infections. Med Princ Pract. 2007;16(5):344–7.
- Sisk BA, Bluebond-Langner M, Wiener L, Mack J, Wolfe J. Prognostic disclosures to children: a historical perspective. Pediatrics. 2016;138(3):e20161278.
- 42. Shukla-Udawatta M, Madani S, Kamat D. An update on pediatric pancreatitis. Pediatr Ann. 2017;46(5):e207–11.
- 43. Balduzzi A, Brivio E, Rovelli A, Rizzari C, Gasperini S, Melzi ML, Conter V, Biondi A. Lessons after the early management of the COVID-19 outbreak in a pediatric transplant and hemato-oncology center embedded within a COVID-19 dedicated hospital in Lombardia, Italy. Estote parati. [Always be prepared]. Bone Marrow Transplant. 2020;55(10):1900–5.

- 44. Horn SR, Stoltzfus KC, Mackley HB, Lehrer EJ, Zhou S, Dandekar SC, et al. Long-term causes of death among pediatric patients with cancer. Cancer. 2020;126(13):3102–13.
- 45. Yanik EL, Smith JM, Shiels MS, Clarke CA, Lynch CF, Kahn AR, et al. Cancer risk after pediatric solid organ transplantation. Pediatrics. 2017;139(5):pii: e20163893.
- Ologbenla A, Hu M, Hajizadeh M. Socioeconomic status and incidence of pediatric leukemia in Canada: 1992-2010. Cancer Epidemiol. 2019;61:14–22.
- 47. Hargreave M, Jensen A, Nielsen TS, Colov EP, Andersen KK, Pinborg A, Kjaer SK. Maternal use of fertility drugs and risk of cancer in children – a nationwide population-based cohort study in Denmark. Int J Cancer. 2015;136(8):1931–9.
- 48. Gilboa D, Koren G, Barer Y, Katz R, Rotem R, Lunenfeld E, Shalev V. Assisted reproductive technology and the risk of pediatric cancer: a population based study and a systematic review and meta analysis. Cancer Epidemiol. 2019;63:101613.
- 49. Johnson KJ, Lee JM, Ahsan K, Padda H, Feng Q, Partap S, Fowler SA, Druley TE. Pediatric cancer risk in association with birth defects: a systematic review. PLoS One. 2017;12(7):e0181246.
- Maese L, Schiffman JD. The evidence for expanded genetic testing for pediatric patients with cancer. Future Oncol. 2018;14(3):187–90.
- Malik P, Edginton A. Pediatric physiology in relation to the pharmacokinetics of monoclonal antibodies. Expert Opin Drug Metab Toxicol. 2018;14(6):585–99.
- 52. Nathan PC, Bremner KE, Liu N, Gupta S, Greenberg ML, McBride ML, et al. Resource utilization and costs in adolescents treated for cancer in pediatric vs adult institutions. J Natl Cancer Inst. 2019;111(3):1–9.

Hyperleukocytosis and Leukostasis

Sharleen Yuan, Samantha J. Yarmis, and Kami M. Hu

Case Study

A 78-year-old male with a history of hypertension, coronary artery disease, and acute myelogenous leukemia is brought in by family for altered mental status. His family members state that he has not acted like himself for the last few days and has had a steady functional decline for the last few weeks to months. Additionally, he has experienced worsening shortness of breath with a fever of 101.2 degrees Fahrenheit (°F) that developed "the other day." The patient is alert but only oriented to self, which is inconsistent with his baseline. He has had no known sick contacts and is up to date on his immunizations. The patient denies any chest pain, upper respiratory symptoms, nausea, vomiting, diarrhea, numbness, tingling, or weakness. He has no history of alcohol or drug use. The rest of his history is unremarkable.

Physical Exam

Vitals: Temp 100.8 °F, BP 150/75 mmHg, HR 98 bpm, RR 32/ min, O_2 saturation 89% on room air. General: Patient alert, but not oriented to place or time. Appears tachypneic when speaking but is in no acute distress. HEENT: Normal sclera, intact extraocular movements, pupils equal and reactive to light. Petechiae noted on buccal mucosa intraorally. Tympanic membranes are normal. Trachea midline. Cardiovascular: Regular rate and rhythm, with good S1 and S2. No murmurs, rubs, or gallops. Pulses 2+ bilaterally. Pulmonary: Rales bilaterally with faint expiratory wheezes

K. M. Hu (🖂)

throughout. Abdomen: Normoactive bowel sounds. Nontender, non-distended, with the spleen tip palpable 3 cm below the costal margin. Neuro: Cranial nerves intact to testing. Motor strength and sensation intact. Negative Romberg, normal gait. Skin: Scattered petechiae over bilateral upper and lower extremities. Extremities: Warm, wellperfused, without edema.

Laboratory Values

White blood cell count, $125 \times 10^3/\mu$ [micro]L (89% blasts); hemoglobin, 6.6 g/dL; hematocrit, 19.9%; platelets, $9 \times 10^3/\mu$ [micro]L; sodium, 137 mmol/L; potassium, 5.3 mmol/L; chloride, 104 mmol/L; bicarbonate, 22 mmol/L; blood urea nitrogen, 25 mg/dL; creatinine, 2.4 μ [micro]mol/L; calcium, 9 mg/dL; magnesium, 2.9 mg/dL; phosphorus, 4 mg/dL; troponin, <0.04 ng/mL; uric acid, 5.5 mg/dL; lactate dehydrogenase, 432 IU/L.

Imaging

Chest radiograph: Bilateral interstitial infiltrates without cardiomegaly, masses or effusions.

Management

The patient was determined to have hyperleukocytosis causing cerebral leukostasis. He received isotonic intravenous (IV) fluids in the emergency department (ED) and was transfused 1 unit of packed red blood cells and 1 pack of platelets. The on-call oncologist was consulted and hydroxyurea was started with a plan for leukapheresis given his altered mental status and worsening WBC count. He was admitted to the cancer service to further monitor his symptoms and was also treated with allopurinol for mild tumor lysis syndrome. By early the next morning, the patient had demonstrated a



S. Yuan · S. J. Yarmis

Department of Emergency Medicine, Department of Internal Medicine, University of Maryland Medical Center, Baltimore, MD, USA e-mail: khu@som.umaryland.edu

Department of Emergency Medicine, Department of Internal Medicine, University of Maryland School of Medicine, Baltimore, MD, USA

mild decrease in WBC count with the hydroxyurea alone, and there were extensive conversations with the patient's family regarding placement of a leukopheresis catheter and overall goals of care. In the end, the family decided on palliative measures and the patient was transitioned to inpatient hospice care.

Introduction

The leukemias are a broad group of hematologic malignancies encompassing acute myeloid leukemia (AML), chronic myeloid leukemia (CML), acute lymphoblastic leukemia/ lymphoma (ALL), and chronic lymphocytic leukemia (CLL) [1]. Given the prevalence of these malignancies, it is highly likely that the average emergency physician will encounter a patient at their first presentation for medical care. This chapter will discuss the ED management of hyperleukocytosis and resulting leukostasis, providing insights into necessary diagnostics and crucial steps for treatment. Other complications of leukemia are discussed elsewhere in the text.

Hyperleukocytosis is most often defined as a white blood cell (WBC) count of greater than 50,000 or 100,000 cells/ microliter, but complications can occur at lower numbers. and absolute WBC count does not correlate directly with symptoms [2]. Although coined specifically to refer to the acute phase of CML, the term "blast crisis" is sometimes used to describe any hyperleukocytosis with predominant blasts. Hyperleukocytosis causes symptoms when the high concentration of circulating cells leads to leukostasis and impaired microvascular circulation causing organ damage, most commonly affecting the lungs and brain. Additional complications of hyperleukocytosis include tumor lysis syndrome (TLS) and disseminated intravascular coagulation (DIC), both independently life-threatening entities. Early recognition and treatment of this disease process is crucial to minimize the morbidity and mortality of these patients and to optimize them for attempts to cure of malignancy.

Epidemiology

Globally, leukemias are the eighth most commonly diagnosed cancer with 600,000 new cases annually and the ninth most common cause of cancer mortality with 353,000 deaths each year [3]. Of these, AML and CLL have the highest annual incidence rates with 190,000 and 191,000 new cases, respectively, while ALL carries the highest yearly mortality with approximately 110,000 deaths annually [3]. The overall incidence of leukemias has increased by 26% from 2005 to 2015, largely due to an aging global population as well as overall population growth [3]. With respect to pediatric patients, leukemias are the most common childhood cancer and the leading cause of both childhood and young adult (age 15–29 years) cancer deaths worldwide [3].

In the United States, leukemias are the 11th most commonly diagnosed cancer [3] with a 1.5% lifetime risk of developing any type of leukemia [4] and the sixth most common cause of cancer deaths each year. While there is a relatively even distribution between acute and chronic leukemia diagnoses annually, when divided by type, CLL is the most commonly diagnosed leukemia, followed by AML, CML, and ALL. The overall number of annual deaths from acute leukemia is two-thirds higher than those from chronic leukemias [5, 6] with AML responsible for the most deaths, followed by CLL, ALL, and CML [3].

Rates of leukemia vary by age and by ethnicity. Overall, acute leukemias are more common in non-Hispanic whites than in other ethnicities [7]. In terms of age, AML demonstrates a bimodal distribution, with a peak in infancy followed by a second peak beginning in young adulthood, with a median age at diagnosis of 69 years [8]. ALL also has a bimodal distribution with a slightly later peak in childhood and then a smaller second peak after the age of 60 [7], with a median age at diagnosis of 15 years. The median age at diagnosis is 65 years for CML and 70 years for CLL [8].

Overall survival for all forms of leukemia has improved, from 33% in 1975 to 59% in 2005 [4], but these gains in survival have been unevenly distributed with the greatest gains for patients with ALL, patients in younger age groups, and patients who are white. For the acute leukemias, the poorest survival is in those under 1 year of age, after which point the likelihood of survival is highest in older children and young adults and subsequently decreases with age [7]. Survival also varies widely by location; childhood ALL, for example, carries an overall survival rate ranging from 15% to 90% depending on the country, reflecting disparities in access to care [3]. As survival from childhood leukemias has improved, treatment-related mortality has accounted for an increased number of deaths, especially from ALL [9]. There are over 450,000 leukemia survivors in the United States [8], with several factors contributing to the increased survival over time. These factors include more intensive and standardized protocols for pediatric patients, the development of tyrosine kinase inhibitors (TKIs) and stem cell transplantation, and improved techniques for diagnosis and residual monitoring, including flow cytometry, polymerase chain reaction (PCR), and fluorescence in situ hybridization (FISH) testing [1, 4, 6].

Epidemiologic data on frequencies of ED presentations by patients with leukemia are scarce. A 2017 analysis of the Nationwide Emergency Department Sample found that of the 29.5 million ED visits by patients with a cancer diagnosis over a 7-year period, 3.1% of those were by patients with a diagnosis of leukemia [10]. The existing published data on general ED usage indicates that solid tumors make up a larger contingency of ED visits [10, 11], but a case series from MD Anderson Cancer Center found that patients with leukemia accounted for the second highest percentage of annual visits by cancer patients (12.7%) with the highest admission rate for a specific cancer (83%) [12].

The incidences of hyperleukocytosis and leukostasis are dependent on the type of leukemia, and the presence or absence of certain genetic mutations affects an individual's risk. Perhaps because myeloid cells are larger than lymphoid cells, leukostasis occurs most often in AML and CML and can occur at lower white blood cell counts in the lymphocytic leukemias [13]. The incidence of hyperleukocytosis in adult patients with AML ranges from 5% to 13%, while it ranges from 10% to 30% for patients with ALL. While hyperleukocytosis is common in CLL, leukostasis itself is rare [14]. Hyperleukocytosis is similarly common in CML, but with more frequent clinical leukostasis occurring most often during blast crisis. Hyperleukocytosis is associated with markedly higher early mortality (approximately 20% in the first week) and modestly reduced 5-year overall survival for those who survive the initial phase [2].

Health Economics

The cost of general cancer care is rising rapidly [12, 15], and the management of leukemia is no exception. There are multiple drivers of healthcare expenditures in treating leukemia, including the cost of medications, the costs associated with the provision of both clinic and in-hospital care, and the costs related to the management of both expected and unexpected treatment-related complications [15, 16]. Treatment strategies for leukemia have expanded to include targeted oral chemotherapies and immunotherapies which often carry decreased treatment-related morbidity and mortality and the potential for increased longevity. These medications are often much more expensive than traditional chemotherapy, however, and are a significant driver of the increased costs of modern leukemia management, as exemplified by the 300% increase in per-patient lifetime cost of CLL treatment since the development of newer oral targeted therapies [15].

Treatment costs vary by leukemia subtype and type of treatment. Costs to the healthcare system range from approximately \$88,000 per chronic leukemia patient to more than \$450,000 per acute leukemia patient over the first year after diagnosis, with highly variable costs to patients themselves [17]. The need for inpatient care is a major driver, accounting for 55–70% of costs in the United States, depending on leukemia type [17–19], and hospitalizations for leukemia cost

more per stay than any other type of cancer [20]. Direct cancer treatment, specifically anticancer drugs and hematopoietic stem cell transplantation (HSCT), accounts for a large portion of healthcare costs [17, 19].

For patients with incurable malignancy or poor prognosis, severe comorbid illness, and poor functional status or who have goals of care that focus on comfort rather than aggressive measures, palliative care consultation is associated with lower rates of intensive care unit (ICU) admission and lower healthcare costs, although such consultations rarely occur early [12, 21, 22].

ED visits are common among leukemia patients; in one study, 81% of patients with acute leukemias had at least one unplanned ED visit and/or hospitalization within the first year [23]. Common reasons for ED visits include fever, bleeding, gastrointestinal complaints, and respiratory symptoms [18]. Robust data on the cost of these unplanned visits as a portion of total leukemia healthcare expenditures is lacking.

Pathophysiology of Leukemia and Leukostasis

Both acute and chronic leukemias arise from abnormalities in hematopoiesis that lead to the proliferation of abnormal cells. The pathophysiology of each will be briefly discussed with the acknowledgement that the processes are complex and not yet fully understood. The increased availability of molecular techniques has led to recognition of greater heterogeneity among the leukemia subtypes than was previously known, and different subtypes of each disease may have different causal mechanisms [18].

In normal hematopoiesis, pluripotent hematopoietic stem cells in the bone marrow give rise to all hematologic cells, with various biochemical factors influencing their differentiation into common lymphoid or common myeloid progenitor cells [24]. Common lymphoid precursors mature within the lymphoid organs to become natural killer cells, T lymphocytes, and B lymphocytes [25], while common myeloid progenitor cells eventually give rise to basophils, neutrophils, eosinophils, macrophage, platelets, and erythrocytes [24]. Blast cells are the early manifestations of both lymphoid and myeloid cells. In healthy individuals, they comprise less than 5% of the cells in the bone marrow and should not be found in the peripheral blood.

In CML, there is abnormal proliferation of the myeloid lineage. The initial chronic phase, characterized by the accumulation of myeloid progenitors and mature myeloid cells in the blood and extramedullary tissues, is followed after 3 to 4 years by arrest of myeloid cell maturation and a transition to blast crisis [26]. Blast crisis is technically defined as the presence of greater than 20% blasts in the blood or bone

marrow with demonstration of extramedullary blast accumulation [1, 27, 28]. In over 90% of cases, patients have a characteristic acquired genetic abnormality, the Philadelphia chromosome, which is the result of translocation between the long arms of chromosomes 9 and 22 [t(9;22)], resulting in the creation of the BCR-ABL fusion gene [26].

AML, as its name suggests, does not have the initial indolent phase seen in CML. It is characterized by uncontrolled clonal expansion of myeloid progenitor cells, i.e., blasts. The accumulation of blasts interferes with the production of normal blood cells of all lineages – red blood cells, platelets, and normal white blood cells – leading to the common signs and symptoms of fatigue, easy bleeding and bruising, and fever and infection, secondary to anemia, thrombocytopenia, and neutropenia, respectively [16, 29]. Acute promyelocytic leukemia (APL, APML), an important subtype of AML with its own prognosis, complications, and specific treatment considerations, accounts for 5–10% of cases and is characterized by the translocation t(15;17) [30]. Several other mutations and cytogenetic abnormalities play key roles in recent targeted therapies [31].

A related malignancy, myelodysplastic syndrome (MDS), is characterized by cytopenias resulting from abnormal, ineffective bone marrow [24, 32]. An enormous variety of genetic mutations have been implicated in the development of MDS, with varying effects on prognosis [32]. Some subtypes of MDS are associated with excess, but fewer than 20%, blasts. The presence of greater than 20% blasts defines AML, and, in fact, having MDS dramatically increases the risk of developing AML [1, 33].

ALL occurs due to the clonal expansion of lymphoid progenitor cells which proliferate and accumulate in the bone marrow and other tissues [34]. Different subtypes are characterized by the type of lymphocytes involved, i.e., B or T cells, as well as the specific causative cytogenetic abnormalities, which have important prognostic and therapeutic implications. A complex interplay between multiple genetic mutations and environmental factors ultimately results in the development of ALL [35].

CLL is characterized by the accumulation of matureappearing B lymphocytes in the blood, bone marrow, or other lymphoid tissues [36]. As in ALL, multiple incompletely understood genetic mutations and other factors result in CLL and determine severity and prognosis [37].

As already mentioned, hyperleukocytosis can cause a hyperviscous state and decreased tissue perfusion, although the fact that particularly high WBC counts may not always cause symptoms is still not fully explained. Theories regarding specific pathophysiology include the consideration that progenitor cells (blasts) are generally larger and less flexible than mature cells, making them more likely to block microvasculature. It has also been proposed that the rapid division of malignant cells, potentially in combination with the body's response to microvascular ischemia, releases specific cytokines and chemoattractants that further attract blast cells and activate the coagulation cascade, worsening the overall prothrombotic milieu [13]. The ongoing cell turnover of such a high volume of cells often leads to a pre-chemotherapy "spontaneous" tumor lysis syndrome. This turnover is also thought to increase the circulating concentration of tissue factor, a procoagulant, with ongoing endothelial damage and hypoxic events predisposing to the development of disseminated intravascular coagulation [2, 13].

Presentation

Leukostasis can be difficult to recognize, as its clinical manifestations mimic other common conditions such as infection, intoxication, acute cerebrovascular accident, and myocardial infarction [2]. The most common presenting symptoms are those of respiratory or neurologic involvement. Patients with pulmonary complaints may present with symptoms ranging from mild dyspnea on exertion to respiratory distress and hypoxia requiring mechanical ventilation. The pulmonary exam, similarly, may range from relatively normal to demonstrating focal findings or widespread abnormalities. Neurologic symptoms can range from focal neurologic deficits to generalized confusion or weakness, blurred vision, or decreased alertness, and a fundoscopic examination may reveal retinal hemorrhages or papilledema. Despite an elevated white count, patients with hyperleukocytosis actually have a functional neutropenia due to decrease in the amount of normal white blood cells, and up to 80% of patients with leukostasis develop fevers, most likely due to increased inflammation as well as high risk of concurrent infection given the patient's immunosuppressed state [28]. The electrolyte abnormalities associated with TLS may cause patients to present with arrhythmia, palpitations, or [pre]syncope, although these are not classic presentations of hyperleukocytosis. Leukostasis can affect any part of the body; the emergency physician should be wary of other organ system dysfunction such as renal insufficiency or myocardial ischemia and perform a thorough physical exam and appropriate diagnostics once hyperleukocytosis is identified [38]. Some common manifestations of leukostasis are listed in Fig. 36.1.

ED Evaluation

Because of the nonspecific nature of these patients' presenting symptoms, the diagnosis of leukemia and leukostasis are often not considered until bloodwork returns with an elevated WBC count, but once hyperleukocytosis is recognized, it is important to revisit the patient's presentation. Ensuring a thorough history is key, utilizing collateral from chart review Fig. 36.1 Common manifestations of leukostasis



CLINICAL MANIFESTATIONS

and available family or friends if the patient is unable to provide it. It may also be necessary to reexamine the patient to ensure all signs of potential organ dysfunction have been identified.

The gold standard diagnostic test for leukostasis is tissue biopsy demonstrating leukocyte-clogged blood vessels [39]. but this test is rarely performed. The diagnosis is now most often made based on a combination of clinical findings and diagnostics indicative of organ dysfunction in the setting of hyperleukocytosis. Laboratory testing in the ED should include a complete blood count (CBC) with differential, peripheral blood smear, complete metabolic panel (CMP) including magnesium and phosphorus levels, prothrombin time with international normalized ratio (PT/INR), partial thromboplastin time (PTT), fibrinogen, D-dimer, fibrin-split products, uric acid, lactate dehydrogenase (LDH), and type and screen. A troponin should be collected in patients with chest discomfort or other concern for myocardial ischemia, and a blood gas obtained for pH determination in patients with metabolic acidosis and acute renal failure. A venous blood gas (VBG) is usually the most appropriate as arterial blood gas (ABG) oxygenation values are inaccurate due to ongoing oxygen consumption by the many leukocytes in the blood collection tube [28]. This "pseudohypoxemia" can be partially mitigated by sending the ABG to the lab on ice. Blood cultures should be obtained in febrile patients, with urine culture added if urinalysis is consistent with infection and sputum culture if patient has a productive cough. An electrocardiogram should be obtained to evaluate for signs of leukostasis-induced ischemia or other pathology secondary to metabolic abnormalities.

Imaging should target the symptoms exhibited by the patient. In patients with respiratory complaints, a chest x-ray should be obtained, and if clear, a non-contrast computed tomography (CT) scan of the chest should be pursued. Chest imaging may demonstrate a wide range of abnormalities, including interstitial edema, consolidation, focal or diffuse airspace opacities, and effusion or other infiltrates [40]. While malignancy is generally associated with an increased risk of venous thromboembolism, patients presenting with new leukemia and hyperleukocytosis prior to treatment are much more likely to have pulmonary leukostasis (reported incidence of 30-34% in AML) [41, 42] than pulmonary embolism (PE) (reported incidence of 3.4% to 5% for all leukemias) [43]. Indiscriminate evaluation by contrasted CT angiography should therefore be avoided if possible, given the high risk of renal dysfunction in this population. Although specific leukemias, such as CML and APL, carry higher risks for the development of venous thromboemboli (VTE), they are most often treatment-related, whether associated with the presence of a central venous catheter or the chemotherapy used [43, 44]. In cases of legitimately high concern for PE, CT must be pursued as ventilation-perfusion (VQ) scans have limited ability to discriminate between VQ mismatch caused by leukostasis versus embolism [45]. A non-contrast CT of the head should be included if there are neurologic symptoms, and it may demonstrate ischemia, hemorrhage, or intracranial mass [46–48]. Depending on the neurologic deficits present, a nondiagnostic CT may indicate a need for further evaluation with magnetic resonance imaging (MRI). The presence of abdominal tenderness on exam may warrant imaging of the abdomen, whether by ultrasound or CT.

ED Management

The mainstay of ED leukostasis management is supportive therapy adhering to the ABCs – or CABs – of emergency medical care, as well as management of concomitant TLS or DIC, if present. While the basics of TLS and DIC treatment will be touched upon here, more thorough discussion can be found elsewhere in the text, and a summary of treatments can be found in Fig. 36.2.

Circulation

Hypotension is not classically seen in this patient population unless they also have severe metabolic acidosis or shock due to another etiology, but isotonic crystalloid IV fluids are a key component of leukostasis treatment for the purpose of hemodilution to decrease blood viscosity. Intravenous fluid administration also helps manage concomitant TLS and preserve kidney function. In patients without a reduced ejection fraction, it is recommended that IV fluids be given to a target urine output of 2 ml/kg/hr [49]. The use of furosemide may be considered if urine output falls below 100 ml/hr, in order to allow continued IV fluid

Fig. 36.2 Summary of ED treatments for leukostasis management. DIC = diffuse intravascular coagulation. *Red blood cell transfusion for hemorrhagic shock only, until white blood cell count decreased



administration, but should be used with caution as diuretics can worsen hyperviscosity and are relatively contraindicated in hypotension unrelated to cardiac volume overload. If the blood pressure is not adequately supported with hydration, a vasopressor such as norepinephrine should be used to maintain a mean arterial pressure (MAP) of greater than 65 mmHg. In patients with decompensated heart failure, use of an inotrope such as epinephrine is more appropriate, keeping in mind that at higher doses (above 0.05 mcg/kg/min) there is more alpha-1-receptor effect and pulmonary vasoconstriction may ensue, worsening potential cardiopulmonary instability.

There should be high suspicion for DIC in patients experiencing hemorrhage, whether pulmonary, intracranial, or another source. If the patient is not bleeding or is hemodynamically stable with small volume blood loss, aggressive packed red blood cell transfusion is not recommended due to its likelihood of worsening hyperviscosity. In patients without hemorrhage, early platelet transfusions should be performed to maintain platelet counts greater than 20,000 μ [micro]L to avoid spontaneous bleeding [50]. If the patient has CNS or intraocular hemorrhage, the platelet goal should be greater than 100,000 µ[micro]L, with a goal of 50,000 µ[micro]L for bleeding at other sites. Patients with DIC and fibrinogen levels less than 150 mg/dL should receive cryoprecipitate transfusion [51].

Patients with acute renal failure, whether secondary to renal ischemia from leukostasis, the renotoxic effects of lysed cell contents, or a combination of both, may exhibit a severe metabolic acidosis depending on the severity of renal injury, with secondary vasodilation and hypotension [52]. Consultation to the nephrology team may be indicated for emergent renal replacement therapy if there is not marked improvement with IV fluids, but the patient with hemodynamic instability and metabolic acidosis with a pH less than 7.2 should be temporized with IV sodium bicarbonate infusion [53] in addition to support with vasopressors if needed.

Airway/Breathing

The most acutely pressing issue for patients with leukostasis can be mild hypoxia or respiratory failure, both of which should be managed with the appropriate respiratory support from simple nasal cannula up to and including intubation if necessary. Patients may also present with increased work of breathing without pulmonary leukostasis secondary to respiratory compensation for metabolic acidosis. Strong consideration should be given to intubation in these patients, in order to decrease their metabolic workload and prevent decompensation when their respiratory muscles fatigue. It is crucial to remember to match their minute ventilation while still paralyzed post-intubation to avoid loss of their compensatory alkalosis and subsequent hemodynamic decompensation.

Other Treatment Considerations

Patients with leukostasis and TLS, characterized by hyperuricemia, hyperkalemia, hyperphosphatemia, and hypocalcemia with or without acute kidney injury, should receive allopurinol, dosed appropriately according to renal function, in addition to IV fluids. If uric acid levels are high, typically greater than 8 mg/dL, a single dose of rasburicase, an effective uric acid-lowering agent [54], may be given in the ED. The electrolyte derangements should be managed as usual, with severe hyperkalemia sometimes requiring emergent dialysis if unresponsive to standard measures. The hypocalcemia that may be seen in parallel to hyperphosphatemia is due to phosphate binding with calcium and, as such, should not be repleted unless the patient is symptomatic or the hyperphosphatemia has been corrected, as IV calcium may cause intravascular calcium-phosphate crystal precipitation and renal failure.

Emergency physicians should maintain a low threshold to obtain cultures and administer empiric antibiotics in patients with hyperleukocytosis and fever, as they are functionally neutropenic [28]. Although these patients are in a hypercoagulable state, without a clear diagnosis of life-threatening venous thromboembolism, prophylactic antiplatelet and anticoagulant medications should not be given due to the patient's high risk of bleeding. It is recommended that patients with identified TLS or DIC or who are unstable should have a CBC, BMP with magnesium and phosphorus, uric acid and coagulation panel, including fibrinogen, with consideration given to pH monitoring as well, measured every 4 to 6 hours while in the ED [55].

Definitive Treatment

The definitive management of hyperleukocytosis and leukostasis is leukocytoreduction, which can be achieved by three methods: administration of hydroxyurea, leukapheresis, and initiation of appropriate chemotherapy. Emergent consultation with the oncologist is crucial to determine the plan for cytoreduction. While hydroxyurea is generally widely available and effective and in most instances should be started immediately, it can take 24 to 48 hours to demonstrate its cytoreductive effect. On the other hand, leukapheresis is rapidly effective but in most cases requires central venous access and is not performed at all centers, potentially requiring transfer to another facility. If the patient is symptomatic, emergent leukapheresis for leukocytoreduction is the treatment of choice, although there is no consistent data to indicate that it improves overall mortality [13]. The emergency physician should note that if placement of a central line is required during their care of the patient, it is helpful to avoid the right internal jugular in case leukapheresis or emergent hemodialysis is indicated, as it is the preferred site for apheresis/dialysis catheter placement.

Disposition and Prognosis

Leukostasis is a medical emergency. Without immediate recognition and treatment, a patient presenting with new onset leukemia and leukostasis with both pulmonary and neurologic symptoms has a 1-week mortality rate of 90% [56]. Pulmonary involvement is an independent predictor of hospital mortality and has been identified as the single worst prognosticator for overall morbidity and mortality [41]. Although other predictors of in-hospital death include altered mental status and non-white race [12], overall mortality of individuals with leukostasis is 40%. A diagnosis of leukostasis requires admission to an inpatient team; the level of care is determined by the patient's clinical stability and hospital resources and protocols.

In consultation with the oncologist, the management of stable, asymptomatic patients sent to the ED for lab abnormalities who exhibit hyperleukocytosis without leukostasis is often less "exciting." These patients can be started on hydroxyurea and IV fluids without much fanfare, depending on the level of hyperleukocytosis, and admitted to the inpatient service. Stable tumor lysis syndrome without lifethreatening electrolyte derangements or metabolic acidosis can similarly be started on allopurinol and/or rasburicase with IV hydration and appropriate lab monitoring. The presence of DIC, even without hemorrhage, warrants closer monitoring as serious bleeding can develop quickly.

For patients with severe medical comorbidities or poor baseline functional status, initiation of palliative services in the ED may be warranted. An increasing number of cancer patients die in the hospital, and leukemia has the second highest in-hospital mortality rate at 16%. Early initiation of palliative care, especially in the ED, has been associated with improved patient symptom management, decreased utilization of hospital resources, and decreased hospital costs [12, 21, 22, 57, 58].

Future Research

Most of the research regarding economic impact and ED utilization is small-scale and limited with respect to the leukemias specifically. Further research into cost and number of and reasons for ED visits, with and without subsequent hospitalization, would be helpful in defining the scope of the problem, assessing changes over time and investigating interventions to decrease healthcare costs and ED utilization.

Only 2–4% of palliative care consults for cancer patients take place in the ED, and consultation often occurs close to the time of death [12]. Research into factors leading to late palliative consultation and factors influencing ED referrals could have a significant impact on the quality of life of leukemia patients, as well as avoiding unnecessary and/or unwanted hospitalizations in these patients. The development of predictive models to determine hospital mortality for patients with leukemia, which could be applied in the ED, could be useful for both patients and physicians.

Several subgroups of patients have been underrepresented in large clinical trials, which may contribute to disparities in care. These include racial minorities and elderly patients [8, 15, 18]. As patient populations continue to increase in age and medical complexity over time, more specific research into both curative and palliative treatments for these sicker patients is needed.

Conclusion

The collective leukemias are some of the most common malignancies worldwide, resulting in more than half a million new cases yearly. Leukostasis is a medical emergency that must be promptly recognized by the emergency physician and is commonly accompanied by other potentially lifethreatening disease processes such as tumor lysis syndrome and disseminated intravascular coagulation. Leukostasis presents nonspecifically and may mimic other disorders. High clinical suspicion is warranted if the patient has a history of hematologic malignancy, and the diagnosis should be quickly identified once hyperleukocytosis is detected. Patients with leukostasis not only require supportive care but also directed therapies to improve outcomes, making it crucial for emergency clinicians to not only be aware of this process but to also know how to take the appropriate steps to manage it.

Pearls

- Leukostasis refers to the clinical effects of blood hyperviscosity and impaired tissue perfusion secondary to hyperleukocytosis and can present with a broad range of symptoms, most often respiratory or neurologic.
- The mainstays of leukostasis management include hemodilution with IV fluids, supportive therapy including respiratory and hemodynamic support if needed, leukocytoreduction with hydroxyurea +/- leukapheresis, and ultimately chemotherapy.
- Pulse oximetry is more accurate than ABGs for assessment of oxygenation, as there is continued oxygen consumption by the leukocytes in the ABG syringe.
- Patients with hyperleukocytosis should be screened for TLS and DIC, which increase mortality and should be quickly addressed. If the patient is unstable and remains in the ED for an extended period, labs should be obtained every 4 to 6 hours.
- Packed red blood cell transfusion should be avoided in relatively stable patients as it worsens viscosity and can further impair perfusion.
- Hydroxyurea is indicated even for asymptomatic or mild symptoms, starting at 50–100 mg/kg/day until WBC decreases appropriately.

References

- Arber DA, Orazi A, Hasserjian R, Thiele J, Borowitz MJ, Le Beau MM, et al. The 2016 revision to the World Health Organization classification of myeloid neoplasms and acute leukemia. Blood. 2016;127(20):2391–405.
- Röllig C, Ehninger G. How I treat hyperleukocytosis in acute myeloid leukemia. Blood. 2015;125(21):3246–52.
- Fitzmaurice C, Allen C, Barber RM, Barregard L, Bhutta ZA, Brenner H, et al. Global, regional, and national cancer incidence, mortality, years of life lost, years lived with disability, and disability-adjusted life-years for 32 cancer groups, 1990 to 2015: a systematic analysis for the global burden of disease study. JAMA Oncol. 2017;3(4):524–48.
- Bailey C, Richardson LC, Allemani C, Bonaventure A, Harewood R, Moore AR, et al. Adult leukemia survival trends in the United States by subtype: a population-based registry study of 370,994 patients diagnosed during 1995–2009. Cancer. 2018;124(19):3856–67.
- Nabhan C, Rosen ST. Chronic lymphocytic leukemia: a clinical review. JAMA. 2014;312(21):2265–76.
- Radich JP, Deininger M, Abboud CN, Altman JK, Berman E, Bhatia R, et al. Chronic Myeloid Leukemia, version 1.2019, NCCN clini-

cal practice guidelines in oncology. J Natl Compr Cancer Netw. 2018;16(9):1108–35.

- Dores GM, Devesa SS, Curtis RE, Linet MS, Morton LM. Acute leukemia incidence and patient survival among children and adults in the United States, 2001-2007. Blood. 2012;119(1):34–43.
- Miller KD, Nogueira L, Mariotto AB, Rowland JH, Yabroff KR, Alfano CM, et al. Cancer treatment and survivorship statistics, 2019. CA Cancer J Clin. 2019;69(5):363–85.
- Hough R, Vora A. Crisis management in the treatment of childhood acute lymphoblastic leukemia: putting right what can go wrong (emergency complications of disease and treatment). Hematology Am Soc Hematol Educ Program. 2017;2017(1):251–8.
- Rivera DR, Gallicchio L, Brown J, Liu B, Kyriacou DN, Shelburne N. Trends in adult cancer–related emergency department utilization: an analysis of data from the nationwide emergency department sample. JAMA Oncol. 2017;3(10):e172450.
- Mayer DK, Travers D, Wyss A, Leak A, Waller A. Why do patients with cancer visit emergency departments? Results of a 2008 population study in North Carolina. J Clin Oncol. 2683;29(19):2011.
- Elsayem AF, Merriman KW, Gonzalez CE, Yeung S-CJ, Chaftari PS, Reyes-Gibby C, et al. Presenting symptoms in the emergency department as predictors of intensive care unit admissions and hospital mortality in a comprehensive cancer center. J Oncol Pract. 2016;12(5):e554–e63.
- Ganzel C, Becker J, Mintz PD, Lazarus HM, Rowe JM. Hyperleukocytosis, leukostasis and leukapheresis: practice management. Blood Rev. 2012;26(3):117–22.
- Porcu P, Cripe LD, Ng EW, Bhatia S, Danielson CM, Orazi A, et al. Hyperleukocytic leukemias and leukostasis: a review of pathophysiology, clinical presentation and management. Leuk Lymphoma. 2000;39(1–2):1–18.
- Chen Q, Jain N, Ayer T, Wierda WG, Flowers CR, O'Brien SM, et al. Economic burden of chronic lymphocytic leukemia in the era of oral targeted therapies in the United States. J Clin Oncol. 2017;35(2):166.
- Bosshard R, O'Reilly K, Ralston S, Chadda S, Cork D. Systematic reviews of economic burden and health-related quality of life in patients with acute myeloid leukemia. Cancer Treat Rev. 2018;69:224–32.
- 17. Gabriela Dieguez F, Ferro C, Rotter CD. The cost burden of blood cancer care. Blood. 2018.
- Wiese M, Daver N. Unmet clinical needs and economic burden of disease in the treatment landscape of acute myeloid leukemia. Am J Manag Care. 2018;24(16 Suppl):S347–S55.
- Pandya BJ, Chen C-C, Medeiros BC, McGuiness CB, Wilson SD, Walsh EH, et al. Economic and clinical burden of acute myeloid leukemia episodes of care in the United States: a retrospective analysis of a commercial payer database. J Manag Care Spec Pharm. 2020;26(7):849–59.
- Price RA, Stranges E, Elixhauser A. Cancer hospitalizations for adults, 2009: Statistical Brief #125. In: Healthcare cost and utilization project (HCUP) statistical briefs [Internet]. Rockville: Agency for Healthcare Research and Quality; 2012. https://www.ncbi.nlm. nih.gov/books/NBK92614/.
- May P, Normand C, Cassel JB, Del Fabbro E, Fine RL, Menz R, et al. Economics of palliative care for hospitalized adults with serious illness: a meta-analysis. JAMA Intern Med. 2018;178(6):820–9.
- 22. May P, Garrido MM, Cassel JB, Kelley AS, Meier DE, Normand C, et al. Cost analysis of a prospective multi-site cohort study of palliative care consultation teams for adults with advanced cancer: where do cost-savings come from? Palliat Med. 2017;31(4):378–86.
- Bryant AL, Deal AM, Walton A, Wood WA, Muss H, Mayer DK. Use of ED and hospital services for patients with acute leukemia after induction therapy: one year follow-up. Leuk Res. 2015;39(4):406–10.

- Meier B, Burton JH. Myeloproliferative disorders. Emerg Med Clin. 2014;32(3):597–612.
- Rose-Inman H, Kuehl D. Acute leukemia. Hematol Oncol Clin. 2017;31(6):1011–28.
- Shet A, Jahagirdar B, Verfaillie C. Chronic myelogenous leukemia: mechanisms underlying disease progression. Leukemia. 2002;16(8):1402.
- Hehlmann R, Saussele S. Treatment of chronic myeloid leukemia in blast crisis. Haematologica. 2008;93(12):1765–9.
- Assi R, Short N. Blast crisis of chronic myeloid leukemia (CML). In: Nates J, Price K, editors. Oncologic critical care. Cham: Springer; 2020. p. 1135–45. https://doi.org/10.1007/978-3-319-74588-6_93.
- DiNardo CD, Cortes JE. Mutations in AML: prognostic and therapeutic implications. Hematol Am Soc Hematol Educ Program. 2016;2016(1):348–55.
- Kantarjian HM, Keating MJ, Freireich EJ. Toward the potential cure of leukemias in the next decade. Cancer. 2018;124(22):4301–13.
- Medinger M, Passweg JR. Acute myeloid leukaemia genomics. Br J Haematol. 2017;179(4):530–42.
- Haferlach T. The molecular pathology of myelodysplastic syndrome. Pathobiology. 2019;86(1):24–9.
- Hasserjian RP. Myelodysplastic syndrome updated. Pathobiology. 2019;86(1):7–13.
- Jabbour E, O'Brien S, Konopleva M, Kantarjian H. New insights into the pathophysiology and therapy of adult acute lymphoblastic leukemia. Cancer. 2015;121(15):2517–28.
- Terwilliger T, Abdul-Hay M. Acute lymphoblastic leukemia: a comprehensive review and 2017 update. Blood Cancer J. 2017;7(6):e577.
- Zenz T, Mertens D, Küppers R, Döhner H, Stilgenbauer S. From pathogenesis to treatment of chronic lymphocytic leukaemia. Nat Rev Cancer. 2010;10(1):37–50.
- Ferrer G, Montserrat E. Critical molecular pathways in CLL therapy. Mol Med. 2018;24(1):9.
- Emerson G, Kaide CG. Rapid fire: acute blast crisis/hyperviscosity syndrome. Emerg Med Clin North Am. 2018;36(3):603–8.
- McCurdy MT, Wacker DA. Selected oncologic emergencies. In: Walls R, Hockberger R, Gausche-Hill M, editors. Rosen's emergency medicine: concepts and clinical practice, vol. 2. 9th ed. Philadelphia: Elsevier; 2018.
- Stefanski M, Jamis-Dow C, Bayerl M, Desai RJ, Claxton DF, Van de Louw A. Chest radiographic and CT findings in hyperleukocytic acute myeloid leukemia: a retrospective cohort study of 73 patients. Medicine (Baltimore). 2016;95(44):e5285.
- 41. Lester TJ, Johnson JW, Cuttner J. Pulmonary leukostasis as the single worst prognostic factor in patients with acute myelocytic leukemia and hyperleukocytosis. Am J Med. 1985;79(1):43–8.
- 42. Moreau A-S, Lengline E, Seguin A, Lemiale V, Canet E, Raffoux E, et al. Respiratory events at the earliest phase of acute myeloid leukemia. Leuk Lymphoma. 2014;55(11):2556–63.
- 43. De Stefano V, Sora F, Rossi E, Chiusolo P, Laurenti L, Fianchi L, et al. The risk of thrombosis in patients with acute leukemia: occurrence of thrombosis at diagnosis and during treatment. J Thromb Haemost. 2005;3(9):1985–92.
- 44. Wu Y-Y, Tang L, Wang M-H. Leukemia and risk of venous thromboembolism: a meta-analysis and systematic review of 144 studies comprising 162,126 patients. Sci Rep. 2017;7(1):1–10.
- Kaminsky DA, Hurwitz CG, Olmstead JI. Pulmonary leukostasis mimicking pulmonary embolism. Leukemia Res. 2000;24(2):175–8.
- Ali AM, Mirrakhimov AE, Abboud CN, Cashen AF. Leukostasis in adult acute hyperleukocytic leukemia: a clinician's digest. Hematol Oncol. 2016;34(2):69–78.
- 47. Algharras AA, Mamourian A, Coyne T, Mohan S. Leukostasis in an adult with AML presenting as multiple high attenuation brain masses on CT. J Clin Diagn Res. 2013;7(12):3020.

- Chim CS, Ooi C. The irreplaceable image: cerebral leukostasis manifesting as multifocal intracerebral hemorrhage. Haematologica. 2001;86(11):1231.
- 49. Mirrakhimov AE, Voore P, Khan M, Ali AM. Tumor lysis syndrome: a clinical review. World J Crit Care Med. 2015;4(2):130.
- Squizzato A, Hunt BJ, Kinasewitz GT, Wada H, Ten Cate H, Thachil J, et al. Supportive management strategies for disseminated intravascular coagulation. Thromb Haemost. 2016;115(5):896–904.
- Spahn DR, Bouillon B, Cerny V, Coats TJ, Duranteau J, Fernández-Mondéjar E, et al. Management of bleeding and coagulopathy following major trauma: an updated European guideline. Crit Care. 2013;17(2):R76.
- Mitchell JH, Wildenthal K, Johnson RL Jr. The effects of acid-base disturbances on cardiovascular and pulmonary function. Kidney Int. 1972;1(5):375–89.
- 53. Jaber S, Paugam C, Futier E, Lefrant J-Y, Lasocki S, Lescot T, et al. Sodium bicarbonate therapy for patients with severe metabolic acidaemia in the intensive care unit (BICAR-ICU): a multicentre, open-label, randomised controlled, phase 3 trial. Lancet. 2018;392(10141):31–40.

- 54. Cortes J, Moore JO, Maziarz RT, Wetzler M, Craig M, Matous J, et al. Control of plasma uric acid in adults at risk for tumor lysis syndrome: efficacy and safety of rasburicase alone and rasburicase followed by allopurinol compared with allopurinol alone—results of a multicenter phase III study. J Clin Oncol. 2010;28(27):4207.
- 55. Coiffier B, Altman A, Pui C-H, Younes A, Cairo MS. Guidelines for the management of pediatric and adult tumor lysis syndrome: an evidence-based review. J Clin Oncol. 2008;26(16):2767–78.
- Porcu P, Farag S, Marcucci G, Cataland SR, Kennedy MS, Bissell M. Leukocytoreduction for acute leukemia. Ther Apher. 2002;6(1):15–23.
- 57. Cheung MC, Earle CC, Rangrej J, Ho TH, Liu N, Barbera L, et al. Impact of aggressive management and palliative care on cancer costs in the final month of life. Cancer. 2015;121(18):3307–15.
- Grudzen CR, Richardson LD, Johnson PN, Hu M, Wang B, Ortiz JM, et al. Emergency department–initiated palliative care in advanced cancer: a randomized clinical trial. JAMA Oncol. 2016;2(5):591–8.

Bleeding and Thrombosis

Thomas G. DeLoughery

Case Study

A 67-year-old man was receiving gemcitabine for metastatic pancreatic cancer. He has noticed after the fourth cycle of treatment his blood pressure was starting to rise. When he presented for his sixth cycle of therapy, his oncologist noted that his platelets were decreased to 50,000/uL. Review of the blood smear showed 2–3 schistocytes per high-power field. His creatinine had risen to 2.3 mg/dL and LDH was five times normal. A diagnosis of gemcitabine-induced thrombotic microangiopathy was made. Despite holding gemcitabine, his laboratory findings continued to worsen. Anti-complement therapy with eculizumab was started with a loading dose of 900 mg weekly and then 1200 mg every other week. Within 2 weeks, his platelets started to rise and his LDH fell. By 2 months, his creatinine had normalized. Given complete resolution of his thrombotic microangiopathy, his anti-complement therapy was stopped. At 1-year follow-up, he had no signs of recurrence of his thrombotic microangiopathy.

Introduction

Cancer patients can suffer both bleeding and thrombotic complications, which can impair both quality of life and survival. These complications can be due to the effects of the cancer itself or its therapy. This chapter will review these complications and offer guidance to diagnosis and treatment.

Bleeding

Bleeding Related to Coagulation Factors

Acquired von Willebrand Disease Acquired von Willebrand disease (VWD) can complicate hematological malignancies – lymphomas, myeloproliferative neoplasms, myeloma, and monoclonal gammopathies [1]. Acquired VWD should be considered when a patient with these types of tumors presents with excess bleeding – especially nosebleeds or gastrointestinal bleeding [2]. Both type 1 (decreased total von Willebrand protein) and type 2 (loss of high molecular weight multimers) VWD can be seen.

Patients with acquired VWD have variable responses to therapy for acute bleeding [3]. Desmopressin is effective for patients with acquired VWD types 1 and 2, but the magnitude and duration of effect are often reduced. For bleeding or for surgical procedures, high doses of the von Willebrand concentrate Humate-P or recombinant von Willebrand factor are indicated with careful monitoring of levels. For patients with very strong inhibitors that factor concentrates cannot overcome or those with life-threatening bleeding, rVIIa may prove useful.

Acquired Factor VIII Inhibitors Factor VIII deficiency is the most frequent acquired coagulation factor deficiency seen in cancer patients [4]. There are also cases reported after use of checkpoint inhibitors [5, 6]. Patients will have prolonged aPTTs, a positive screening test for a factor inhibitor, and a low factor VIII level. For severe or life-threatening bleeding, recombinant VIIa is the treatment of choice [7]. The dose is 90 ug/kg repeated every 2–3 h until bleeding has stopped. The key step beside controlling bleeding is to eliminate the autoantibody with immunosuppression starting with prednisone 60 mg/day and adding oral cyclophosphamide 100 mg po plus rituximab either 375 mg/m² weekly × 4 or 1000 mg separated by 14 days. Eradication of the autoantibody does not require successful tumor treatment

[©] Springer Nature Switzerland AG 2021 K. H. Todd et al. (eds.), *Oncologic Emergency Medicine*, https://doi.org/10.1007/978-3-030-67123-5_37

T. G. DeLoughery (🖂)

Division of Hematology/Medical Oncology, Department of Medicine, Oregon Health & Science University, Portland, OR, USA e-mail: delought@ohsu.edu

and should be attempted before major procedures such as tumor resection are planned.

Bleeding Related to Platelet Number and Function

Immune Thrombocytopenia Immune thrombocytopenic purpura (ITP) has been reported in 2-4% of patients with chronic lymphocytic leukemia, myeloma, and Hodgkin's disease [8]. ITP can occur during any part of the course of the tumor including when the patient has responded to antineoplastic therapy. The presentation of ITP in cancer patients is not different than in those with other conditions; patients present with petechia and other stigmata of bleeding. Why ITP occurs during the course of lymphoproliferative disease is not well understood, although a disturbed immune system could possibly predispose patients both to the lymphoproliferative disorder and to thrombocytopenia [9, 10]. Increasingly, ITP and other autoimmune cytopenias are being reported as complications of the use of checkpoint inhibitors [11]. Therapy for the ITP that complicates tumors/ checkpoint therapy is the same as that for classic immune thrombocytopenia. High-dose corticosteroids such as pulse dexamethasone 40 mg/day × 4 days (plus immunoglobulin 1 gram/kg IV if necessary) are given first, but if an adequate platelet count cannot be maintained, then the choice is splenectomy, thrombopoietin between agonists, and rituximab.

Thrombotic Thrombocytopenic Purpura Thrombotic thrombocytopenic purpura (TTP) should be suspected when a patient presents thrombocytopenia, microangiopathic hemolytic anemia (schistocytes and signs of hemolysis), and any evidence of end-organ damage [12]. Diagnosis is confirmed by finding a low ADAMT13 level (<10%). TTP has a unique presentation in cancer patients with evidence of metastatic cancer in the bone marrow and lungs [13, 14]. These patients can have extensive intravascular tumor leading to thrombocytopenia and schistocytes. Plasma exchange is ineffective in these patients, but resolution of the TTP has been reported if the tumor is responsive to antineoplastic therapy.

Therapy-Related Thrombotic Microangiopathy Thrombotic microangiopathies (TM) can complicate a variety of therapies such as calcineurin inhibitors, proteasome inhibitors [15], and gemcitabine [16]. The most common antineoplastic drug causing TM is gemcitabine with an incidence of 0.1-1% [17, 18]. The appearance of the TM syndrome associated with gemcitabine can be delayed, and the condition often is fatal. Severe hypertension often precedes the clinical appearance of the TM. The use of plasma exchange is waning due to lack of effectiveness, and

there are increasing reports of the successful use of the complement inhibitor eculizumab [19].

TMs can complicate stem cell marrow transplants [20, 21]. The incidence ranges from 15% for allogeneic to 5% for autologous bone marrow transplants. Several types of TMs are recognized in bone marrow transplantations. One is "multi-organ fulminant" which occurs early (20-60 days) with the evolution of multi-organ system involvement and is often fatal. Another type of TM is similar to calcineurin inhibitors TMs. This occurs within days after the agent is started with the appearance of a falling platelet count, falling hematocrit, and rising serum LDH level [22]. With withdrawing the agent, the TM resolves. A "conditioning" TM, which occurs 6 months or more after total body irradiation, is associated with primary renal involvement. Finally, patients with systemic CMV infections can present with a TM syndrome related to vascular infection with CMV. The etiology of bone marrow transplant-related TM appears to be different from that of "classic" TTP since alterations of ADAMTS13 have not been found in stem cell transplant-related TTP implicated therapy-related vascular damage. There is increasing evidence that disorders of complement regulation may be the etiology of early stem cell transplant TM. Although plasma exchange is often tried, response is poor for TTP/HUS. There are increasing reports of successful use of the complement inhibitor eculizumab, and this agent should be used, especially if there are signs of complement dysregulation such as high C5-9 levels [23, 24].

Specific Hematological Cancers Associated with Bleeding

Acute Promyelocytic Leukemia (APL)

Patients with APL have a higher risk of hemorrhagic death during induction therapy when compared with patients with other forms of leukemia [25]. The hemostatic defects in patients with APL are multiple with most having evidence of DIC at the time of diagnosis with bleeding still being the major cause of early death [26]. Life-threatening bleeding such as intracranial hemorrhage may occur at any time until the APL is put into remission. The etiology of the hemostatic defects in APL is complex and is thought to be the result of DIC, fibrinolysis, and the release of other procoagulant enzymes [25-27]. Therapy of APL involves treating both the leukemia and the coagulopathy (Table 37.1). Currently the standard treatment for APL is trans-retinoic acid (ATRA) in combination with chemotherapy or arsenic [28, 29]. This will induce remission in over 90% of patients, and a sizable majority of these patients will be cured of their APL. ATRA

Table 37.1 Acute promyelocytic leukemia

Consider diagnosis in any patient with leukemia presenting with coagulopathy Start empiric all-trans-retinoic acid 45 mg/m²/day in two divided doses while performing work-up Coagulation goals Fibrinogen greater than 150 mg/dL Platelets greater than 50 × 10⁹/L

therapy will also lead to the early correction of the coagulation defects, often within the first week of therapy. Given the marked beneficial effect of ATRA on the coagulopathy of APL and its low toxicity, it should be empirically started for any patient suspected of having APL, while specific testing is being performed. Therapy for the coagulation defects consists of aggressive transfusion to maintain the fibrinogen level at over 150 mg/dL and the platelet count at over $50 \times 109/L$.

Myeloproliferative Neoplasms

Bleeding can be seen in many of the myeloproliferative neoplasms, but rarely results in major morbidity [30, 31]. In patients with essential thrombocytosis, the risk of bleeding appears to increase with platelet counts above one million perhaps due to the large amount of platelets absorbing von Willebrand factor. Most bleeding in myeloproliferative neoplasms consists of platelet-type bleeding - mucocutaneous bleeding or bruising with only a few reports of major bleeding. The use of drugs that inhibit platelet function such as aspirin is associated with a higher incidence of bleeding. Patients with extreme thrombocytosis and bleeding will respond to lowering the counts to below $1000 \times 109/L$ [1]. Patients with myeloproliferative neoplasms should be screened for VWD before surgery or starting antiplatelet therapy. Rare patients with myeloproliferative neoplasms will have an acquired factor V deficiency with symptomatic patients presenting with bleeding and variable elevation of the INR and/or aPTT [32, 33].

Dysproteinemia

Multiple coagulation abnormalities have been described in patients with dysproteinemia, which can lead to severe bleeding [34, 35]:

- Abnormal clot retraction
- Abnormal fibrin clot
- Anti-glycoprotein IIb/IIIa antibodies
- Factor VIII inhibitor
- Heparin-like anticoagulation

- · Impaired fibrin polymerization
- Inhibition of thrombin time

Therapy for the hemostatic defects in the dysproteinemic syndromes includes removal of the offending protein, either by reducing synthesis by treating the myeloma with aggressive chemotherapy or by plasmapheresis if the patient is having acute symptoms.

Patients with amyloidosis can have a marked increase in easy bruising and other bleeding symptoms which may be the first clue to diagnosis [36]. The most common defect is an elevation in the thrombin time which is seen in 30–80% of cases. A rare but important cause of bleeding in patients is systemic fibrinolysis. The patients may have decreased levels of alpha-2-antiplasmin and an abnormal euglobulin clot lysis time. The use of fibrinolytic inhibitors such as tranexamic acid has both corrected laboratory tests of fibrinolysis and reduced bleeding symptoms. Like with myeloma, treatment of the amyloid will correct the bleeding diathesis.

Coagulation Defect Due to Therapy

Bleeding has been reported with the use of tyrosine kinase inhibitors used in CML therapy, but it's unclear if this is due to the drug effect or the underlying disease. Many of these agents have been reported to lead to in vitro platelet dysfunction, but for most patients this does not appear to be clinically significant [37]. Bleeding has also been reported with inhibitors of VEGF [38]. Bevacizumab in particular is associated with bleeding - especially after surgery or with treatment of squamous cell carcinoma of the lung. In these cases, the bleeding may be more related to lack of wound healing and tumor necrosis. The BTK inhibitor ibrutinib is associated with a 5% incidence of bleeding including subdural hematoma, with the mechanism appearing to be via decreased platelet aggregation [39-41]. The new BTK inhibitor acalabrutinib - being more selective - may be associated with less bleeding [42].

Cancer and Thrombosis

Epidemiology

Thrombosis can be the presenting sign of cancer [43, 44]. As many as 10–20% of older patients who present with an idiopathic deep venous thrombosis will be found to have cancer on initial evaluation. Furthermore, over the next 2 years, up to 25% of these patients will develop cancer. Certain presentations of thrombosis are more worrisome for underlying cancer as a cause of the thrombosis: warfarin-refractory thrombosis, idiopathic bilateral deep vein thrombosis, or both arterial and venous thrombosis. The cancers most frequently associated with thrombosis are adenocarcinoma of the lung and gastrointestinal cancers, especially pancreatic. Primary brain tumors are also associated with a higher risk of thrombosis as well as kidney, ovarian, and uterine cancers [43]. Thrombosis rates for breast and prostate cancer are not as extreme [45]. The co-existence of cancer and thrombosis has implications for both disease processes [46]. Cancer raises the risk of both anticoagulant-induced bleeding and breakthrough thrombosis, while the presence of a thrombosis worsens the cancer prognosis.

Given the data that thrombosis can be an early sign of cancer, one question that commonly arises is "Should a patient who presents with an idiopathic thrombosis be aggressively worked up for cancer?" Studies to date have not shown benefit of extensive evaluations of these patients for cancer, and current recommendations are age-appropriate cancer screening and complete work-up of any worrisome signs – such as guaiac-positive stools.

Increasingly common in cancer patients is the finding of an "incidental" pulmonary embolism on a CT obtained for tumor staging or evaluation of response to chemotherapy. Despite the "incidental" nature of finding the thrombosis, the prognosis is just as ominous as any cancer-related thrombosis, and these need to be aggressively treated with anticoagulation [47]. Also in cancer patients, distal vein thrombosis has a high recurrence rates and mandates long-term anticoagulation [48].

Rare patients can present with thrombosis and associated disseminated intravascular coagulation. These patients with tumor-related DIC have thrombosis in the setting of low platelets and decrease coagulation factors. These patients may also develop a non-bacterial thrombotic endocarditis and have multiple arterial embolic events.

The etiology of the cancer-related thrombosis is complex with many factors potentially playing a role [49, 50]. Tumors may directly activate factor VII by tumor-expressed tissue factor, as well as factor X. Patients with cancer have elevations of inflammatory cytokines that can further augment the hypercoagulable state. Treatment of the cancer can also lead to thrombosis. As discussed below, chemotherapy – especially cisplatinum, fluorouracil, and asparaginase – increases the risk of thrombosis. Biological agents such as thalidomide and lenalidomide also increase thrombosis risk. Surgery for cancer patients increases the risk of thrombosis threefold over similar operations in non-cancer patients [51].

Treatment

Cancer-related thrombosis requires aggressive anticoagulation [52, 53]. While in the past low molecular weight heparin (LMWH) was the treatment of choice, this has now shifted to the direct oral anticoagulants (DOAC). There are three large clinical trials of LMWH vs DOAC that show that DOACs are T. G. DeLoughery

as effective, if not more effective, in the prevention of recurrent thrombosis with a slighter higher risk of bleeding [54– 56]. This higher bleeding risk is only seen with upper gastrointestinal cancers and not with the use of apixaban. Two clinical trials have also shown the benefit of prophylactic dosing of DOAC by preventing the first cancer thrombosis in higher thrombotic risk patients [57, 58]. However, given the low absolute risk reduction in thrombosis, widespread use remains controversial.

Patients who have recurrent thrombosis while taking direct oral anticoagulants need to be treated indefinitely with LMWH. The rare patient who fails LMWH may benefit either from raising the dose by 25% or changing to fondaparinux [59].

Brain tumors or brain metastases are not a contraindication to anticoagulation. The only exceptions are brain metastases from thyroid cancer, melanoma, renal cancer, or choriocarcinoma, as these tumor metastases have a high rate of bleeding [60]. It should be remembered that placement of an inferior vena cava filter without concurrent anticoagulation is associated with an unacceptable rate of complications, including death from massive thrombosis.

Specific Situations

Myeloproliferative Neoplasms

Thrombosis is the most common cause of death in the myeloproliferative neoplasms [31, 61]. Correlation of thrombosis with blood counts depends on the underlying disease – patients with polycythemia rubra vera are at risk of thrombosis with hematocrits over 45%, but those with essential thrombocytosis can have thrombosis with platelets in the $4-600 \times 109/L$ and as noted before may have greater risk of bleeding with counts greater than $1000 \times 109/L$. Patients with myeloproliferative neoplasms have a higher risk of thrombosis even with relatively normal blood counts, suggesting an intrinsic defect in the blood cell or vascular endothelium leading to thrombosis.

Patients with myeloproliferative neoplasms have a predilection for several types of thrombosis. Patients with Budd-Chiari and other visceral vein thromboses have a high incidence of underlying myeloproliferative syndromes [62]. Patients with essential thrombocytosis can also have platelet occlusion of the small digital vessels leading to erythromelalgia. These patients will have swollen, red, and very painful digits. The patients may only have slightly elevated platelet counts and are often misdiagnosed with arthritis. A diagnostic clue is that these patients will respond dramatically to a single aspirin per day.

Certain patients, especially those with Budd-Chiari syndrome, may have an "occult" myeloproliferative syndrome with maybe no evidence of any hematological disorder but have genetic evidence of myeloproliferative disease with positive testing for the JAK2 mutation. Interestingly, the *CALR* mutation – seen in 30–50% of patients with essential thrombocytosis – has not been associated with visceral vein thrombosis in patients with normal blood counts [63].

Heparin followed by warfarin or direct oral anticoagulants is indicated for most patients with acute venous thromboembolism complicating the myeloproliferative disorders. In a few instances, liver transplantation has been successful in treating liver failure due to Budd-Chiari syndrome – but these patients require long-term anticoagulation.

Antiplatelet therapy is recommended for treatment of patients with arterial thrombosis. Low doses of aspirin (81 mg/d) are preferable in patients with myeloproliferative neoplasms because the risk of bleeding with aspirin is dose related. There is increasing data that dosing 81 mg bid may be more effective [64, 65]. There is currently no data concerning the use of newer agents such as clopidogrel, but it may be reasonable for patients allergic to aspirin.

In addition to antithrombotic therapy, treating elevated blood counts is also important for patients with a history of thrombosis [66, 67]. For patients with thrombocytosis, hydroxyurea (1 gm daily to start) is the preferred therapy as trials have shown antithrombotic benefit. For younger patients who have concerns about hydroxyurea, weekly PEGylated interferon starting at 45–90 ug is another consideration. For patients with polycythemia, reduction of the hematocrit to under 45% with phlebotomy, hydroxyurea, or interferon is crucial [68]. There is increasing data for using the *JAK2* inhibitor ruxolitinib for blood count control in patients intolerant of other therapy, and this is another treatment option with retrospective data showing reduction in thrombosis [69, 70].

A common issue is whether to reduce platelet counts or to give aspirin to patients with thrombocytosis who do not have a history of thrombosis. Platelet reduction with hydroxyurea or interferon should be considered in asymptomatic patient if they are older (>65 years) or they have atherosclerosis, risk factors for arterial disease or symptoms of vascular ischemia. Some studies indicate that an elevated white count or the presence of the *JAK2* mutation may also be a risk factor for thrombosis [71]. All patients diagnosed with polycythemia should have their hematocrits reduced to less than 45% with either phlebotomy or cytoreduction [68]. Aspirin should be used in all patients with polycythemia (unless they have a bleeding diathesis). For patients with thrombocytosis, those at low risk of thrombosis do not benefit from aspirin [72].

Paroxysmal Nocturnal Hemoglobinuria (PNH)

One of the leading causes of morbidity and mortality in patients with PNH is thrombosis with patients presenting with either venous or arterial disease [73, 74]. PNH is also associated with a high incidence of visceral vein thrombosis.

The cause of the hypercoagulable state is unknown, but complement-activated platelets have been implicated. In two large series pre-dating specific anti-compliment therapy, the rate of thrombosis in PNH was 28–39% with thrombosis leading to death in 58% [73, 75]. The development of the complement inhibitor eculizumab has led to control of the hemolysis in most patients with PNH, and there is strong evidence it also reduces thrombosis rates [76]. Eculizumab or ravulizumab [77] should be used in any patient with PNH who has had thrombosis, severe hemolysis, or a significant PNH clone (>50%). Although PNH is rare, patients with visceral vein thrombosis, thrombosis with unexplained high LDH levels, recurrent or warfarin-refractory thrombosis, or thrombosis in the setting of pancytopenia should be screened for PNH.

Venous Catheter Thrombosis

Central venous catheters are essential to many aspects of cancer therapy, but the clinically apparent thrombosis incidence for catheters is estimated to be 3-6% [78]. The signs of catheter thrombosis are non-specific, and the incidence of thrombosis is thought to be underestimated (Table 37.2).

Patients with catheter-related thrombosis often notice arm pain and swelling. Diagnosis of the thrombosis is made by Doppler, but some patients may only have central vein thrombosis and may require venography or CT angiography to make the diagnosis. Many patients have the diagnosis found while undergoing imaging for other reasons.

Therapy is not well defined. Data is increasing for peripherally inserted central catheters that simply removing the catheter may be the safest approach as the risk of bleeding with anticoagulation is high – reserving anticoagulation for the severely symptomatic [79]. For thrombosis with tunneled lines, anticoagulation should be given – unless the risk of bleeding is substantial. One trial does show that one can keep the catheter in place with 3 months of anticoagulation [80]. Prevention of catheter thrombosis is difficult as prophylaxis has not been shown to be a benefit.

 Table 37.2
 Options for catheter thrombosis

Per

Tu

eripheral inserted central catheters
Removal of catheter
Reserve anticoagulation for very symptomatic patient
unneled central catheters
Evaluated if line is needed
If removed, short-term anticoagulation if no bleeding
If kept in place, 3 months of anticoagulation

risk factors

Antineoplastic Therapy

Adjuvant chemotherapy for breast cancer has been associated with an increased risk of both arterial and venous thromboembolism (in 5-7% of patients) [81]. The thrombogenic stimulus is not clear, but this could reflect vascular damage by the chemotherapeutic agents or perhaps a reduction in natural anticoagulants, such as protein C or protein S concentrations.

L-Asparaginase – an effective therapy for acute lymphocytic leukemia – is associated with thrombosis [82, 83]. The overall rate of thrombosis in children is 5% but may be as high as 36% if asymptomatic thromboses are included and can range from 5% to 20% in adult. The rate of potentially devastating CNS thrombosis is approximately 1-2% of patients with childhood ALL and up to 4% of adults. Thrombosis usually occurs 2–3 weeks after the start of a course of therapy. Most patients recover, although serious neurologic defects or even death can occur.

The pathogenesis of the thrombotic complications of L-asparaginase may be related to decreased levels of natural anticoagulants antithrombin III, protein C, protein S, and plasminogen via general inhibition hepatic protein synthesis by L-asparaginase.

Patients with acute thrombosis should have levels of fibrinogen and antithrombin drawn before anticoagulation and if deficient supplemented to keep fibrinogen greater than 150 mg/dL and antithrombin greater than 80%. Platelets need to be kept greater than $50 \times 109/L$ during acute anticoagulation.

There remains no consensus on prevention of thrombosis given varying results of clinical trials. There is increasing evidence that LMWH prophylaxis at either 40 mg/day or 1 mg/kg/day is effective at preventing thrombosis [84].

The anti-myeloma agents thalidomide and lenalidomide are both associated with substantial rates of thrombosis that can be as high as 36–75% [85]. The incidence is higher with the use of dexamethasone and with chemotherapy, especially doxorubicin. These agents may have a direct toxic effect on the vascular endothelium promoting a prothrombotic state. Aspirin appears useful for thrombosis prevention in low-risk patients, while those who have had previous thrombosis, receiving dexamethasone or chemotherapy, or have central lines may benefit from warfarin or LMWH prophylaxis.

Targeted antineoplastic therapy also increases the risk of thrombosis. Bevacizumab has been associated with a ~twofold increase in arterial thrombosis [86] but not venous disease [87]. This may be a class effect of VEGF inhibition as the VEGF tyrosine kinase inhibitors such as sorafenib and sunitinib also increase arterial thrombosis 2.2-fold [88]. Several of the new tyrosine kinase inhibitors developed for treatment of chronic myelogenous leukemia also increase the risk of arterial thrombosis – the most pronounce risk with ponatinib with an incidence ratio of 40.7 per 100 patient years [89, 90].

Use of Anticoagulants in Thrombocytopenic Patients

A common issue for which there is little guidance is management of anticoagulation in patients who are or are at risk of becoming thrombocytopenic [91, 92]. For venous thrombosis, full-dose heparin/DOAC can be given to a platelet count of 50,000/uL and prophylactic dosing down to $20 \times 109/L$ [93]. Aspirin given for primary prevention can be held until therapy is over. For secondary prevention, one would hold aspirin when platelets decrease to under $50 \times 109/L$. For acute coronary events, patients should receive aspirin no matter what their platelet counts are [94].

References

- Eikenboom JC, Tjernberg P, Van Marion V, Heering KJ. Acquired von Willebrand syndrome: diagnostic problems and therapeutic options. Am J Hematol. 2007;82(1):55–8.
- Michiels JJ, Budde U, van der Planken M, van Vliet HH, Schroyens W, Berneman Z. Acquired von Willebrand syndromes: clinical features, aetiology, pathophysiology, classification and management. Best Pract Res Clin Haematol. 2001;14(2):401–36.
- Tiede A, Rand JH, Budde U, Ganser A, Federici AB, How I. treat the acquired von Willebrand syndrome. Blood. 2011;117(25):6777–85.
- Escobar MA. Bleeding in the patient with a malignancy: is it an acquired factor VIII inhibitor? Cancer. 2012;118(2):312–20.
- Gokozan HN, Friedman JD, Schmaier AH, Downes KA, Farah LA, Reeves HM. Acquired hemophilia A after nivolumab therapy in a patient with metastatic squamous cell carcinoma of the lung successfully managed with rituximab. Clin Lung Cancer. 2019;20(5):e560–3.
- Mouri A, Kaira K, Shiono A, Miura YU, Kagamu H. Severe thrombocytopenia associated with pembrolizumab in patients with nonsmall cell lung cancer (NSCLC): a case report and literature review. In Vivo. 2020;34(2):877–80.
- Barnett B, Kruse-Jarres R, Leissinger CA. Current management of acquired factor VIII inhibitors. Curr Opin Hematol. 2008;15(5):451–5.
- Zent CS, Ding W, Reinalda MS, Schwager SM, Hoyer JD, Bowen DA, et al. Autoimmune cytopenia in chronic lymphocytic leukemia/small lymphocytic lymphoma: changes in clinical presentation and prognosis. Leuk Lymphoma. 2009;50(8):1261–8.
- Kirshner JJ, Zamkoff KW, Gottlieb AJ. Idiopathic thrombocytopenic purpura and Hodgkin's disease: report of two cases and a review of the literature. Am J Med Sci. 1980;280(1):21–8.
- Kempin S. Late hematologic complications after treatment of Hodgkin's disease. In: Lacher MJ, Redman JR, editors. Hodgkin's disease: the consequences of survival. Philadelphia: Lea & Febiger; 1990.
- Michot JM, Lazarovici J, Tieu A, Champiat S, Voisin AL, Ebbo M, et al. Haematological immune-related adverse events with immune checkpoint inhibitors, how to manage? Eur J Cancer. 2019;122:72–90.

- Murrin RJ, Murray JA. Thrombotic thrombocytopenic purpura: aetiology, pathophysiology and treatment. Blood Rev. 2006;20(1):51–60.
- Lechner K, Obermeier HL. Cancer-related microangiopathic hemolytic anemia: clinical and laboratory features in 168 reported cases. Medicine (Baltimore). 2012;91(4):195–205.
- 14. Weitz IC. Thrombotic microangiopathy in cancer. Semin Thromb Hemost. 2019;45(4):348–53.
- 15. Nguyen MN, Nayernama A, Jones SC, Kanapuru B, Gormley N, Waldron PE. Proteasome inhibitor-associated thrombotic microangiopathy: a review of cases reported to the FDA adverse event reporting system and published in the literature. Am J Hematol. 2020. https://doi.org/10.1002/ajh.25832. Epub ahead of print.
- Moake JL, Byrnes JJ. Thrombotic microangiopathies associated with drugs and bone marrow transplantation. Hematol Oncol Clin North Am. 1996;10(2):485–97.
- Saif MW, McGee PJ. Hemolytic-uremic syndrome associated with gemcitabine: a case report and review of literature. JOP. 2005;6(4):369–74.
- Fung MC, Storniolo AM, Nguyen B, Arning M, Brookfield W, Vigil J. A review of hemolytic uremic syndrome in patients treated with gemcitabine therapy. Cancer. 1999;85(9):2023–32.
- Al Ustwani O, Lohr J, Dy G, Levea C, Connolly G, Arora P, Iyer R. Eculizumab therapy for gemcitabine induced hemolytic uremic syndrome: case series and concise review. J Gastrointest Oncol. 2014;5(1):E30–3.
- Pettitt AR, Clark RE. Thrombotic microangiopathy following bone marrow transplantation. Bone Marrow Transplant. 1994;14(4):495–504.
- Khosla J, Yeh AC, Spitzer TR, Dey BR. Hematopoietic stem cell transplant-associated thrombotic microangiopathy: current paradigm and novel therapies. Bone Marrow Transplant. 2018;53(2):129–37.
- Gharpure VS, Devine SM, Holland HK, Geller RB, O'Toole K, Wingard JR. Thrombotic thrombocytopenic purpura associated with FK506 following bone marrow transplantation. Bone Marrow Transplant. 1995;16(5):715–6.
- Wanchoo R, Bayer RL, Bassil C, Jhaveri KD. Emerging concepts in hematopoietic stem cell transplantation-associated renal thrombotic microangiopathy and prospects for new treatments. Am J Kidney Dis. 2018;72(6):857–65.
- Jodele S, Dandoy CE, Lane A, Laskin BL, Teusink-Cross A, Myers KC, et al. Complement blockade for TA-TMA: lessons learned from a large pediatric cohort treated with eculizumab. Blood. 2020;135(13):1049–57.
- Choudhry A, DeLoughery TG. Bleeding and thrombosis in acute promyelocytic leukemia. Am J Hematol. 2012;87(6):596–603.
- Arbuthnot C, Wilde JT. Haemostatic problems in acute promyelocytic leukaemia. Blood Rev. 2006;20(6):289–97.
- Kwaan HC, Weiss I, Tallman MS. The role of abnormal hemostasis and fibrinolysis in morbidity and mortality of acute promyelocytic leukemia. Semin Thromb Hemost. 2019;45(6):612–21.
- Lo-Coco F, Avvisati G, Vignetti M, Thiede C, Orlando SM, Iacobelli S, et al.; Gruppo Italiano Malattie Ematologiche dell'Adulto; German-Austrian Acute Myeloid Leukemia Study Group; Study Alliance Leukemia. Retinoic acid and arsenic trioxide for acute promyelocytic leukemia. N Engl J Med. 2013;369(2):111–21.
- Jimenez JJ, Chale RS, Abad AC, Schally AV. Acute promyelocytic leukemia (APL): a review of the literature. Oncotarget. 2020;11(11):992–1003.
- Falanga A, Rickles FR. Management of thrombohemorrhagic syndromes (THS) in hematologic malignancies. Hematology Am Soc Hematol Educ Program. 2007:165–71. https://doi.org/10.1182/ asheducation-2007.1.165.
- Stein BL, Martin K. From Budd-Chiari syndrome to acquired von Willebrand syndrome: thrombosis and bleeding complications in the myeloproliferative neoplasms. Blood. 2019;134(22):1902–11.

- Hasegawa DK, Bennett AJ, Coccia PF, Ramsay NK, Nesbit ME, Krivit W, Edson JR, Factor V. deficiency in Philadelphia-positive chronic myelogenous leukemia. Blood. 1980;56(4):585–95.
- Ashrani AA, Tefferi A, Pruthi RK, Pardanani A. Acquired factor V deficiency in myeloproliferative neoplasms: a Mayo Clinic series of 33 patients. Br J Haematol. 2015;171(5):875–9.
- Coppola A, Tufano A, Di Capua M, Franchini M. Bleeding and thrombosis in multiple myeloma and related plasma cell disorders. Semin Thromb Hemost. 2011;37(8):929–45.
- Zangari M, Elice F, Fink L, Tricot G. Hemostatic dysfunction in paraproteinemias and amyloidosis. Semin Thromb Hemost. 2007;33(4):339–49.
- Eby CS. Bleeding and thrombosis risks in plasma cell dyscrasias. Hematology Am Soc Hematol Educ Program. 2007;2007:158–64.
- Quintás-Cardama A, Han X, Kantarjian H, Cortes J. Tyrosine kinase inhibitor-induced platelet dysfunction in patients with chronic myeloid leukemia. Blood. 2009;114(2):261–3.
- Sonpavde G, Bellmunt J, Schutz F, Choueiri TK. The double edged sword of bleeding and clotting from VEGF inhibition in renal cancer patients. Curr Oncol Rep. 2012;14(4):295–306.
- Wang ML, Rule S, Martin P, Goy A, Auer R, Kahl BS, et al. Targeting BTK with ibrutinib in relapsed or refractory mantle-cell lymphoma. N Engl J Med. 2013;369(6):507–16.
- 40. Levade M, David E, Garcia C, Laurent PA, Cadot S, Michallet AS, et al. Ibrutinib treatment affects collagen and von Willebrand factor-dependent platelet functions. Blood. 2014;124(26):3991–5.
- Shatzel JJ, Olson SR, Tao DL, McCarty OJT, Danilov AV, DeLoughery TG. Ibrutinib-associated bleeding: pathogenesis, management and risk reduction strategies. J Thromb Haemost. 2017;15(5):835–47.
- 42. Tullemans BME, Heemskerk JWM, Kuijpers MJE. Acquired platelet antagonism: off-target antiplatelet effects of malignancy treatment with tyrosine kinase inhibitors. J Thromb Haemost. 2018;16(9):1686–99.
- Connolly GC, Francis CW. Cancer-associated thrombosis. Hematology Am Soc Hematol Educ Program. 2013;2013:684–91.
- 44. Streiff MB. Association between cancer types, cancer treatments, and venous thromboembolism in medical oncology patients. Clin Adv Hematol Oncol. 2013;11(6):349–57.
- 45. Chew HK, Wun T, Harvey D, Zhou H, White RH. Incidence of venous thromboembolism and its effect on survival among patients with common cancers. Arch Intern Med. 2006;166(4):458–64.
- Timp JF, Braekkan SK, Versteeg HH, Cannegieter SC. Epidemiology of cancer-associated venous thrombosis. Blood. 2013;122(10):1712–23.
- O'Connell C. How I treat incidental pulmonary embolism. Blood. 2015;125(12):1877–82; quiz 2009.
- 48. Poudel SK, Park DY, Jia X, Wilks M, Pinkava V, O'Brien M, et al. Clinical outcomes of isolated distal deep vein thrombosis versus proximal venous thromboembolism in cancer patients: The Cleveland Clinic experience. J Thromb Haemost. 2020;18(3):651–9.
- Lee AY. Cancer and thromboembolic disease: pathogenic mechanisms. Cancer Treat Rev. 2002;28(3):137–40.
- 50. Dicke C, Langer F. Pathophysiology of Trousseau's syndrome. Hamostaseologie. 2015;35(1):52–9.
- 51. Orfanakis A, Deloughery T. Patients with disorders of thrombosis and hemostasis. Med Clin North Am. 2013;97(6):1161–80.
- 52. Lyman GH, Bohlke K, Khorana AA, Kuderer NM, Lee AY, Arcelus JI, et al.; American Society of Clinical Oncology. Venous thromboembolism prophylaxis and treatment in patients with cancer: American Society of Clinical Oncology Clinical Practice Guideline Update 2014. J Clin Oncol. 2015;33(6):654–6.
- 53. Kearon C, Akl EA, Comerota AJ, Prandoni P, Bounameaux H, Goldhaber SZ, et al. Antithrombotic therapy for VTE disease: Antithrombotic Therapy and Prevention of Thrombosis, 9th ed: American College of Chest Physicians Evidence-Based

Clinical Practice Guidelines. Chest. 2012;141(2 Suppl):e419S–96S. Erratum in: Chest. 2012;142(6):1698–704.

- 54. Young AM, Marshall A, Thirlwall J, Chapman O, Lokare A, Hill C, et al. Comparison of an oral factor Xa inhibitor with low molecular weight heparin in patients with cancer with venous thromboembolism: results of a randomized trial (SELECT-D). J Clin Oncol. 2018;36(20):2017–23.
- 55. Raskob GE, van Es N, Verhamme P, Carrier M, Di Nisio M, Garcia D, et al.; Hokusai VTE Cancer Investigators. Edoxaban for the treatment of cancer-associated venous thromboembolism. N Engl J Med. 2018;378(7):615–24.
- Agnelli G, Becattini C, Meyer G, Muñoz A, Huisman MV, Connors JM, et al. Caravaggio Investigators. Apixaban for the treatment of venous thromboembolism associated with cancer. N Engl J Med. 2020;382(17):1599–607.
- 57. Carrier M, Abou-Nassar K, Mallick R, Tagalakis V, Shivakumar S, Schattner A, et al.; AVERT Investigators. Apixaban to prevent venous thromboembolism in patients with cancer. N Engl J Med. 2019;380(8):711–9.
- Soff GA, Mones J, Wilkins C, Devlin S, Haegler-Laube E, Wills J, et al. Rivaroxaban treatment of cancer-associated venous thromboembolism: Memorial Sloan Kettering Cancer Center institutional experience. Res Pract Thromb Haemost. 2019;3(3):349–56.
- 59. Carrier M, Le Gal G, Cho R, Tierney S, Rodger M, Lee AY. Dose escalation of low molecular weight heparin to manage recurrent venous thromboembolic events despite systemic anticoagulation in cancer patients. J Thromb Haemost. 2009;7(5):760–5.
- Gerber DE, Grossman SA, Streiff MB. Management of venous thromboembolism in patients with primary and metastatic brain tumors. J Clin Oncol. 2006;24(8):1310–8. Erratum in: J Clin Oncol. 2006;24(13):2133.
- 61. Barbui T, Finazzi G, Falanga A. Myeloproliferative neoplasms and thrombosis. Blood. 2013;122(13):2176–84.
- 62. Sant'Antonio E, Guglielmelli P, Pieri L, Primignani M, Randi ML, Santarossa C, et al. Splanchnic vein thromboses associated with myeloproliferative neoplasms: an international, retrospective study on 518 cases. Am J Hematol. 2020;95(2):156–66.
- Haslam K, Langabeer SE. Incidence of CALR mutations in patients with splanchnic vein thrombosis. Br J Haematol. 2015;168(3):459–60.
- 64. Larsen ML, Pedersen OH, Hvas AM, Niekerk PBVK, Bønløkke S, Kristensen SD, Grove EL. Once- versus twice-daily aspirin treatment in patients with essential thrombocytosis. Platelets. 2019;30(3):322–8.
- 65. Rocca B, Tosetto A, Betti S, Soldati D, Petrucci G, Rossi E, et al. A randomized double-blind trial of 3 aspirin regimens to optimize antiplatelet therapy in essential thrombocythemia. Blood. 2020;136(2):171–82.
- 66. De Stefano V, Carobbio A, Di Lazzaro V, Guglielmelli P, Iurlo A, Finazzi MC, et al. Benefit-risk profile of cytoreductive drugs along with antiplatelet and antithrombotic therapy after transient ischemic attack or ischemic stroke in myeloproliferative neoplasms. Blood Cancer J. 2018;8(3):25.
- 67. De Stefano V, Rossi E, Carobbio A, Ghirardi A, Betti S, Finazzi G, et al. Hydroxyurea prevents arterial and late venous thrombotic recurrences in patients with myeloproliferative neoplasms but fails in the splanchnic venous district. Pooled analysis of 1500 cases. Blood Cancer J. 2018;8(11):112.
- Marchioli R, Finazzi G, Specchia G, Cacciola R, Cavazzina R, Cilloni D, et al. CYTO-PV Collaborative Group. Cardiovascular events and intensity of treatment in polycythemia vera. N Engl J Med. 2013;368(1):22–33.
- 69. Vannucchi AM, Kiladjian JJ, Griesshammer M, Masszi T, Durrant S, Passamonti F, et al. Ruxolitinib versus standard therapy for the treatment of polycythemia vera. N Engl J Med. 2015;372(5):426–35.

- Masciulli A, Ferrari A, Carobbio A, Ghirardi A, Barbui T. Ruxolitinib for the prevention of thrombosis in polycythemia vera: a systematic review and meta-analysis. Blood Adv. 2020;4(2):380–6.
- Passamonti F, Caramazza D, Mora B, Casalone R, Maffioli M. It is time to change thrombosis risk assessment for PV and ET? Best Pract Res Clin Haematol. 2014;27(2):121–7.
- 72. Godfrey AL, Campbell PJ, MacLean C, Buck G, Cook J, Temple J, et al.; United Kingdom Medical Research Council Primary Thrombocythemia-1 Study; United Kingdom National Cancer Research Institute Myeloproliferative Neoplasms Subgroup; French Intergroup of Myeloproliferative Neoplasms; the Australasian Leukaemia and Lymphoma Group. Hydroxycarbamide plus aspirin versus aspirin alone in patients with essential thrombocythemia age 40 to 59 years without high-risk features. J Clin Oncol. 2018;36(34):3361–9.
- 73. Socié G, Mary JY, de Gramont A, Rio B, Leporrier M, Rose C, et al. Paroxysmal nocturnal haemoglobinuria: long-term follow-up and prognostic factors. French Society of Haematology. Lancet. 1996;348(9027):573–7.
- 74. Hill A, DeZern AE, Kinoshita T, Brodsky RA. Paroxysmal nocturnal haemoglobinuria. Nat Rev Dis Primers. 2017;3:17028.
- Hillmen P, Lewis SM, Bessler M, Luzzatto L, Dacie JV. Natural history of paroxysmal nocturnal hemoglobinuria. N Engl J Med. 1995;333(19):1253–8.
- Hillmen P, Young NS, Schubert J, Brodsky RA, Socié G, Muus P, et al. The complement inhibitor eculizumab in paroxysmal nocturnal hemoglobinuria. N Engl J Med. 2006;355(12):1233–43.
- Stern RM, Connell NT. Ravulizumab: a novel C5 inhibitor for the treatment of paroxysmal nocturnal hemoglobinuria. Ther Adv Hematol. 2019;10:2040620719874728.
- Ellis ML, Okano S, McCann A, McDowall A, Van Kuilenburg R, McCarthy AL, et al. Catheter-related thrombosis incidence and risk factors in adult cancer patients with central venous access devices. Intern Med J. 2020. https://doi.org/10.1111/imj.14780. Epub ahead of print.
- 79. Jones MA, Lee DY, Segall JA, Landry GJ, Liem TK, Mitchell EL, Moneta GL. Characterizing resolution of catheter-associated upper extremity deep venous thrombosis. J Vasc Surg. 2010;51(1):108–13.
- 80. Kovacs MJ, Kahn SR, Rodger M, Anderson DR, Andreou R, Mangel JE, et al. A pilot study of central venous catheter survival in cancer patients using low-molecular-weight heparin (dalteparin) and warfarin without catheter removal for the treatment of upper extremity deep vein thrombosis (The Catheter Study). J Thromb Haemost. 2007;5(8):1650–3.
- Rella C, Coviello M, Giotta F, Maiello E, Colavito P, Colangelo D, et al. A prothrombotic state in breast cancer patients treated with adjuvant chemotherapy. Breast Cancer Res Treat. 1996;40(2):151–9.
- Truelove E, Fielding AK, Hunt BJ. The coagulopathy and thrombotic risk associated with L-asparaginase treatment in adults with acute lymphoblastic leukaemia. Leukemia. 2013;27(3):553–9.
- De Stefano V, Za T, Ciminello A, Betti S, Rossi E. Haemostatic alterations induced by treatment with asparaginases and clinical consequences. Thromb Haemost. 2015;113(2):247–61.
- 84. Sibai H, Chen R, Liu X, Falcone U, Schimmer A, Schuh A, et al. Anticoagulation prophylaxis reduces venous thromboembolism rate in adult acute lymphoblastic leukaemia treated with asparaginasebased therapy. Br J Haematol. 2020. https://doi.org/10.1111/ bjh.16695. Epub ahead of print.
- Palumbo A, Rajkumar SV, Dimopoulos MA, Richardson PG, San Miguel J, Barlogie B, et al.; International Myeloma Working Group. Prevention of thalidomide- and lenalidomide-associated thrombosis in myeloma. Leukemia. 2008;22(2):414–23.
- 86. Scappaticci FA, Skillings JR, Holden SN, Gerber HP, Miller K, Kabbinavar F, et al. Arterial thromboembolic events in patients with metastatic carcinoma treated with chemotherapy and beva-

cizumab. J Natl Cancer Inst. 2007;99(16):1232–9. Erratum in: J Natl Cancer Inst. 2008;100(2):156. Erratum in: J Natl Cancer Inst. 2008;100(9):685.

- 87. Hurwitz HI, Saltz LB, Van Cutsem E, Cassidy J, Wiedemann J, Sirzén F, et al. Venous thromboembolic events with chemotherapy plus bevacizumab: a pooled analysis of patients in randomized phase II and III studies. J Clin Oncol. 2011;29(13):1757–64.
- Choueiri TK, Schutz FA, Je Y, Rosenberg JE, Bellmunt J. Risk of arterial thromboembolic events with sunitinib and sorafenib: a systematic review and meta-analysis of clinical trials. J Clin Oncol. 2010;28(13):2280–5.
- 89. Jain P, Kantarjian H, Boddu PC, Nogueras-González GM, Verstovsek S, Garcia-Manero G, et al. Analysis of cardiovascular and arteriothrombotic adverse events in chronic-phase CML patients after frontline TKIs. Blood Adv. 2019;3(6):851–61.

- Valent P, Hadzijusufovic E, Schernthaner GH, Wolf D, Rea D, le Coutre P. Vascular safety issues in CML patients treated with BCR/ ABL1 kinase inhibitors. Blood. 2015;125(6):901–6.
- Deloughery TG. Between Scylla and Charybdis: antithrombotic therapy in hematopoietic progenitor cell transplant patients. Bone Marrow Transplant. 2012;47(10):1269–73.
- Al-Samkari H, Connors JM. Managing the competing risks of thrombosis, bleeding, and anticoagulation in patients with malignancy. Blood Adv. 2019;3(22):3770–9.
- Falanga A, Marchetti M. Venous thromboembolism in the hematologic malignancies. J Clin Oncol. 2009;27(29):4848–57.
- 94. Sarkiss MG, Yusuf SW, Warneke CL, Botz G, Lakkis N, Hirch-Ginsburg C, et al. Impact of aspirin therapy in cancer patients with thrombocytopenia and acute coronary syndromes. Cancer. 2007;109(3):621–7.



Infectious Disease

Nathanial S. Nolan, Michael J. Kim, and Stephen Y. Liang

Introduction

While the emergency physician specializes in evaluating and managing patients with undifferentiated illness, those with cancer and undergoing treatment represent a unique challenge due to their immunocompromised state, lack of typical inflammatory signs and symptoms, and atypical clinical presentations. In such cases, fever may be the only presenting symptom of infection; however, even lack of fever cannot preclude an infectious disease emergency. Malignancies and their treatment modify immune defenses in many ways, and frequent healthcare interventions, including central venous catheterization, further add to the risk of infection. Cytotoxic chemotherapy often results in disruption of mucosal barriers and reduction in functional neutrophils, both of which are critical components of the innate immune system.

Febrile neutropenia in a patient with cancer is an oncologic emergency. While many advances have been made in prevention and management of febrile neutropenia syndromes, infection remains a serious cause of morbidity and mortality in oncologic patients. Major complications (e.g., hypotension, acute renal failure, respiratory failure) in the context of febrile neutropenia occur at a rate of approximately 25–30% with mortality rates as high as 11%; in the setting of severe sepsis or septic shock, mortality can approach 50% [1–3]. Rapid assessment and timely interventions, including the initiation of empiric antibiotics, can be crucial in preventing complications and deterioration in a patient's clinical course. Therefore, emergency physicians must maintain a high index of suspicion for infection in any oncologic patient presenting to the emergency department (ED) with acute illness.

Risk Factors and Immune Compromise

Oncologic patients have many risk factors for infection. Malignancy frequently causes immune compromise, particularly those malignancies affecting the hematopoietic system. Cancers can be inflammatory, releasing cytokines resulting in disordered immune responses. Primary or metastatic tumors can result in anatomic defects promoting infection, such as an endobronchial lesion predisposing a patient to post-obstructive pneumonia. Furthermore, many cancer treatments result in unintended adverse effects, including neutropenia and mucositis. Surgical interventions and procedures, such as port placement, can be complicated by subsequent infection. Due to immune impairment, many oncologic patients may lack classic symptoms of infection (e.g., fever, signs of inflammation). Therefore, a strong clinical suspicion for infection should be maintained during assessment of the acutely ill oncologic patient presenting for ED care.

The spectrum of infectious diseases seen in oncologic patients can be broadly categorized based on primary immune dysfunction (Table 38.1) [4, 5]. For example, patients with primary neutropenia are most susceptible to infection from bacteria they are already colonized with or those found in their environment. These patients frequently develop bacteremia from gastrointestinal pathogens (due to gut translocation), oral pathogens (due to mucositis), or skin flora (due to skin breakdown and/or sites of central venous access). They are also at high risk for infection due to fungi, including *Candida, Aspergillus*, and other molds [4].

Patients who have lymphomas, those receiving T-celltargeting therapies, or those taking high-dose steroids are considered to have dysfunction of cellular-mediated

N. S. Nolan

Division of Infectious Disease, Department of Medicine, Barnes Jewish Hospital/Washington University School of Medicine, St. Louis, MO, USA

M. J. Kim

Division of Emergency Medicine, Department of Medicine, Barnes Jewish Hospital/Washington University School of Medicine, St. Louis, MO, USA

S.Y. Liang (🖂)

Divisions of Emergency Medicine and Infectious Diseases, Department of Medicine, Washington University School of Medicine, St. Louis, MO, USA e-mail: syliang@wustl.edu

K. H. Todd et al. (eds.), Oncologic Emergency Medicine, https://doi.org/10.1007/978-3-030-67123-5_38

mmune		
dysfunction	Associated conditions	At-risk pathogens
Neutrophil defects	Cytotoxic chemotherapy Hematologic malignancy Marrow infiltration Drug-induced neutropenia	Bacteria: Staphylococcus aureus, Viridans streptococci, Pseudomonas, Stenotrophomonas, Enterobacteriaceae, Clostridium Fungi: Candida, Aspergillus, Fusarium, Zygomycetes, Trichosporon
Cellular immunity defects	Stem cell transplantation Chronic immunosuppressive therapy Hodgkin's lymphoma HIV infection Irradiation	Bacteria: Mycobacteria, atypical bacteria (Nocardia, Legionella, Listeria), zoonoses (Bartonella, Brucella) Fungi: endemic mycoses (Histoplasma, Blastomyces, Coccidioides, Paracoccidioides, Penicillium), Pneumocystis jirovecii, Cryptococcus, Candida, Aspergillus Viruses: herpesviruses (CMV, HSV, VZV, EBV, HHV-6), HPV, community- acquired viral infections Parasites: Toxoplasma, Strongyloides, Leishmania, Cryptosporidium
Humoral immunity defects	Drug-induced B-lymphocyte depletion Splenectomy or functional asplenia Hematopoietic stem cell transplantation Paraproteinemias Non-Hodgkin's lymphoma	Bacteria: Streptococcus pneumoniae, Neisseria meningitidis, Haemophilus influenzae, Capnocytophaga (particularly if receiving rituximab or alemtuzumab) Viruses: Enteroviruses Parasites: Giardia, Plasmodium, Babesia

 Table 38.1
 Immune defects and their associated pathogens [5]

CMV cytomegalovirus, HSV herpes simplex virus, VZV varicella-zoster virus [VZV], EBV Epstein-Barr virus, HHV human herpes virus

immunity. They are particularly vulnerable to infection due to mycobacteria and atypical bacteria, such as Nocardia, Legionella, and Listeria, as well as viruses, including herpesviruses (i.e., cytomegalovirus [CMV], herpes simplex virus [HSV], varicella-zoster virus [VZV], and Epstein-Barr virus [EBV]). They are also more susceptible to infection due to fungi, including dimorphic molds such as Histoplasma, Coccidioides, and Blastomyces, as well as Pneumocystis jirovecii, Aspergillus, and Cryptococcus [4]. Finally, those with therapies targeting B-cells (e.g., monoclonal antibodies against CD-20 surface proteins), hypogammaglobulinemia secondary to malignancy, or those post-splenectomy are considered to have deficits of humoral immunity, placing them at increased risk for infection due to encapsulated bacteria including Streptococcus pneumoniae, Haemophilus influenzae, and Neisseria meningitidis. These patients also have an increased risk of parasitic infection.

Antimicrobial Prophylaxis for Cancer Patients

Prophylactic antimicrobial, antifungal, and antiviral therapies are often employed to mitigate the risk of infection associated with necessary and life-saving cancer treatments. Prophylaxis is tailored to an individual patient's risks and is frequently left to the discretion of the oncologic provider. Hospitals and healthcare systems often have their own internal policies or guidelines, and it is useful to be familiar with these. Herein are the most common oncologic scenarios in which patients receive antimicrobial prophylaxis and the underlying reasoning for doing so. Understanding the rationale for commonly prescribed antimicrobial prophylaxis can help emergency physicians better navigate the differential diagnosis of infection in an acutely ill oncologic patient.

Hematopoietic Stem Cell Transplantation (HSCT)

HSCT is a life-saving procedure for patients with hematologic malignancy. However, this treatment often comes with risk of infection from a variety of pathogens. The process of HSCT (whether allogeneic or autologous) requires a cytotoxic conditioning regimen that produces a period of time in which the patient does not have a completely functional immune system. The degree of immunosuppression will depend on the conditioning regimen and the time it takes to achieve engraftment of stem cells. Further immune compromise is common in allogeneic HSCT patients who suffer from graft-vs-host disease (GVHD). The immunosuppression required to treat this condition often further impairs an already fragile immune system.

The role for prophylaxis in HSCT is well established [6, 7]. Prior to HSCT, the recipient and donor often undergo screening as a means to risk stratify and tailor post-transplant antimicrobial prophylaxis for the recipient. For example, a recipient with existing IgG antibodies to CMV will receive antiviral prophylaxis with a drug that is active against CMV, such as valganciclovir, as opposed to prophylaxis with acyclovir, which targets only HSV and VZV. Likewise, a prior history of invasive mold infection can prompt initiation of mold active antifungal prophylaxis. In general, most prophylaxis is aimed at bacteria commonly found on mucosal surfaces or the gastrointestinal tract (e.g., Gram-negative rods including *Pseudomonas aeruginosa*), fungi (e.g., *Candida*, *Aspergillus*), and herpesviruses. Pathogens with

the greatest risk

Antibacterial

prophylaxis

Antifungal

prophylaxis

prophylaxis

Antiviral

probial prophylaxis in the HSCT patient				
Pre-engraftment phase	Early post-engraftment phase	Late post-engraftment phase		
Gram-negative bacilli (including	Gram-negative bacilli (including	Encapsulated bacteria, VZV,		
Pseudomonas), Gram- positive cocci,	Pseudomonas), Gram-positive cocci, HSV,	Aspergillus, Pneumocystis		
HSV, Candida spp., Aspergillus	CMV, Candida spp., Aspergillus	jirovecii		
Fluoroquinolones (ciprofloxacin or	Penicillin, amoxicillin, or	Penicillin, amoxicillin, or		

trimethoprim-sulfamethoxazole

trimethoprim-sulfamethoxazole

Valacyclovir vs valganciclovir

Fluconazole,

Tab	le 38.2	Antimicrobial	prophylaxis	in the	HSCT	patient
-----	---------	---------------	-------------	--------	------	---------

levofloxacin)

valganciclovir

Fluconazole; if increased risk for mold

Valacyclovir; if CMV IgG positive,

infection, voriconazole, or posaconazole

HSCT hematopoietic stem cell transplantation CMV cytomegalovirus virus, HSV herpes simplex virus, VZV varicella-zoster virus

Infectious risk for patients is often dependent on their phase of recovery following HSCT (Table 38.2). In the preengraftment phase immediately post-transplantation, typically the first 30 days, patients are in various stages of engrafting new stem cells. Patients are generally profoundly neutropenic and often have mucosal barrier disruption. Studies have suggested that antimicrobial prophylaxis with fluoroquinolones (ciprofloxacin or levofloxacin) can reduce the rate of infections with Gram-negative organisms [8, 9]. During this neutropenic phase, HSCT patients also have a risk of invasive fungal infection, most notably with Candida. Antifungal prophylaxis is often targeted toward *Candida*. consisting of daily fluconazole. However, some institutions and/or protocols may favor the use of antifungals such as voriconazole or posaconazole with expanded mold coverage depending on the type of transplant and expected duration of neutropenia. Antiviral prophylaxis is typically directed against human herpesviruses.

A brief discussion of risk for CMV reactivation is worthwhile, as this may seem counterintuitive in HSCT. A highrisk patient is a recipient who is CMV IgG positive at the time of transplant [10]. This high-risk status exists regardless of the donor's status; however, a stem cell transplant from a CMV IgG-positive donor may be protective. As CMV infection is typically a disease of reactivation in HSCT, transplantation from a seronegative donor into a seropositive recipient effectively provides a high-risk recipient with an immune system that has not previously encountered CMV. Various strategies for the management of CMV reactivation exist. Some institutions surveille for reactivation and then provide treatment as needed. Frequently, institutions provide preemptive CMV prophylaxis based on recipient seropositive status [11]. Valganciclovir [12] and letermovir [13] have been demonstrated to reduce the risk of CMV disease in seropositive HSCT patients.

Engraftment of the transplant is evident when the patient is able to maintain a neutrophil count from their own hematopoiesis. Despite the development of improved (and sometimes normal) neutrophil counts, HSCT patients remain immunosuppressed due to impaired activity of B-cells and T-cells.

This impaired immune function can last for years in allogeneic HSCT patients. The early post-engraftment phase is often arbitrarily defined as the time of engraftment around day 30 until day 100. Following engraftment, antimicrobial prophylaxis targets encapsulated bacteria, particularly S. pneumoniae [6]. Based on local protocols, patients may continue on fluoroquinolones, trimethoprim-sulfamethoxazole, or penicillin. The benefit of trimethoprim-sulfamethoxazole is that it also serves as prophylaxis against P. jirovecii, a pathogen known to affect HSCT patients [14]. P. jirovecii prophylaxis is generally started after engraftment due to concern for bone marrow suppression with trimethoprim-sulfamethoxazole. Antifungal prophylaxis against Candida is typically continued during this phase for up to 75–100 days [15].

The late post-engraftment phase refers to the period of time from 100 days until 1-year post-transplant. In this phase, cellular immunity remains impaired. Antibiotic prophylaxis against encapsulated bacteria, specifically S. pneumoniae, is typically continued for up to a year. There is evidence to support the role of prophylaxis against HSV and VZV for up to a year to prevent infection [16]. Currently, CMV prophylaxis is not recommended in the late postengraftment phase [17].

Graft-vs-host disease (GVHD) occurs when transplanted immune cells (from the stem cell transplant) react to recipient tissues. Manifestations of GVHD vary among HSCT recipients. Treatment of GVHD often consists of steroids and frequently escalates to involve immunosuppression with JAK inhibitors or calcineurin inhibitors. In patients with GVHD, prophylaxis often intensifies based on patient risk factors. Some institutions transition antifungal coverage to include mold coverage, if the patient was not already on this. There is some data to suggest that in GVHD, posaconazole and, theoretically, voriconazole decrease the risk for invasive aspergillosis and death [18]. Antibiotic prophylaxis is often continued and will include trimethoprim-sulfamethoxazole for P. jirovecii prophylaxis. Antivirals will continue as previously prescribed. In patients with GVHD, antimicrobial prophylaxis often continues until immunosuppression is decreased.

trimethoprim-sulfamethoxazole

institutions continue for 1 year

Valacyclovir - many

Chemotherapy-Induced Neutropenia

Cytotoxic chemotherapy is a frequent cause of bone marrow suppression and subsequent neutropenia. The risk of infection associated with neutropenia is related to the severity (i.e., how low the neutrophil count is) and the duration of neutropenia. Severe neutropenia is defined as an absolute neutrophil count less than 500 cells/ μ L. Patients with severe neutropenia for a duration of more than 7 days are considered to be high risk for infection.

Recent guidelines suggest that prophylaxis for neutropenia should be targeted and based on the patient's risk for neutropenia-related infection [19]. For patients with risk of severe neutropenia, current evidence suggests benefit with fluoroquinolone antibiotic prophylaxis [20, 21]. Levofloxacin is often preferred due to its spectrum of activity against Streptococci and P. aeruginosa. Ciprofloxacin may also be considered. Antifungal prophylaxis with fluconazole, or other triazoles, is recommended in patients who are expected to have severe neutropenia [19]. Prophylaxis against P. jirovecii, with trimethoprim-sulfamethoxazole, is also recommended if the risk of *Pneumocystis* pneumonia is felt to be high (typically based on use of high-dose corticosteroids or purine analog therapy) [19]. Of note, use of trimethoprimsulfamethoxazole alone as prophylaxis against Pneumocystis and bacterial pathogens is not recommended due to their limited spectrum of activity compared to fluoroquinolones.

Chemotherapy and Immunotherapy Against Solid Organ Tumors

Though chemotherapy for treatment of solid organ tumors risks depressing the immune system, routine prophylaxis is generally not given. Based on current guidelines, the decision to use fluoroquinolones for prophylaxis in solid organ tumor patients receiving cytotoxic chemotherapy with severe neutropenia should be made on a case-by-case basis [22]. In patients who will be treated with high-dose steroids (\geq 20 mg of prednisone daily for \geq 4 weeks), consideration should be given to *P. jirove-cii* prophylaxis using trimethoprim-sulfamethoxazole. Newer immune system-modulating therapies used in cancer treatments may promote risk for invasive infections [23]. Adverse effects of these medications are frequently treated with immune-suppressing therapies (such as corticosteroids). At this time, no guidance on routine prophylaxis before their use exists.

Evaluation of the Oncologic Patient with Infection

When approaching care of the acutely ill oncologic patient in the ED, providers must decipher signs and symptoms that frequently overlap infectious and malignant syndromes. While a fever in an oncologic patient usually indicates infection, it may also result from medications, venous thromboembolism, or the malignancy itself [24].

Fever can be a paraneoplastic syndrome in a wide range of tumor types, from hematologic cancers such as lymphoma and leukemia to solid cancers such as renal cell carcinoma, glioblastoma multiforme, and ovarian carcinoma [25]. While the exact mechanisms of how malignancy induces fever are not fully understood, it is theorized that the release of pyrogenic cytokines directly from tumor cells, or from macrophages responding to the tumor, can induce prostaglandin E2, which acts on the hypothalamus and causes a change in the thermostatic set point. These cytokines - IL-1, IL-2, IL-6, IL-12, TNF, and interferon - also play a crucial role in driving the inflammatory response to infection [25, 26]. Other theories have focused on activation of cytokines by mutated receptors such as RAS; and IL-6 levels are often associated with driving outcomes in lymphomas and renal cell carcinoma [27]. Because the inflammatory states of both neoplastic and infectious origins share similar cytokines, clinical presentations may be indistinguishable between the two, requiring a careful and thorough workup to differentiate them. However, there are differences in the clinical features of fever caused by malignancy compared to infection (Fig. 38.1). Systemic signs and symptoms such as rigors, chills, tachycardia, or hypotension are often muted with malignancy compared to infectious fever. Further, neoplastic fevers are often not relieved by acetaminophen [27].

When approaching a fever of unclear etiology, the first priority of the emergency physician is to treat any presumed underly-



Fig. 38.1 A comparison of neoplastic and infectious fever

ing infection. A fever of unclear etiology in an oncologic patient should be treated as infectious until proven otherwise based on careful examination, laboratory data, and imaging. Neoplastic fever should only be considered once other diagnoses have been thoroughly excluded. The evolving story of SARS-CoV-2 is covered in a separate chapter of this text.

Febrile Neutropenia: An Oncologic Emergency

A frequently encountered syndrome in patients receiving cytotoxic chemotherapy (and sometimes in patients with hematologic malignancies) is neutropenic fever, which is considered an oncologic emergency. This is defined as a temperature \geq 38.3 °C once or \geq 38.0 °C sustained over an hour in the setting of absolute neutrophil count (ANC) <500 cells/µL or an ANC <1000 with predicted nadir <500 cells/µL [28]. Neutropenic fever requires prompt evaluation and empiric antimicrobial treatment as indicated. The choice of antibiotics and location of treatment is based on underlying risk factors [22, 28]. Patients with comorbid conditions, those with severe neutropenia (ANC <100 cells/µL), and those with neutropenia that will last \geq 7 days are at greatest risk for febrile neutropenia.

History and Physical Examination

A careful review of the patient's history and a thorough physical examination should be undertaken. When evaluating an oncologic patient in the ED, it is helpful to sort out the following information:

- Location of the tumor and/or metastases with attention to potential anatomic changes as a result (e.g., does the patient have metastases that could cause urinary, respiratory, bowel, or biliary obstruction?)
- What foreign objects does the patient have inside them (e.g., do they have a central venous catheter, implantable port, urinary catheter, drain, or other implantable device?)
- What therapies are being used to treat the patient and when did he/she last receive them? (e.g., did the patient receive outpatient chemotherapy or are they receiving immunotherapy?)
- What infections has the patient had and/or what infections are they currently being treated for? (e.g., does the patient have a history of a multidrug-resistant organism? Are they currently receiving antimicrobial prophylaxis?)
- What other comorbidities (e.g., diabetes, end-stage renal disease) does the patient have that may increase their risk for infection or worsen clinical outcomes?

While these points do not constitute all of the history that should be obtained, they can help narrow the differential diagnosis. A thorough physical examination can also elucidate supporting signs and symptoms for a suspected infection. Assessing the patient's overall clinical appearance can be the first step in establishing a clinical gestalt - does the patient appear sick or unwell? Has there been a change from their baseline mental status? Central venous catheters, implantable ports, and surgical sites should be inspected for any signs of infection that could lead to bacteremia or wound infection. A tender, distended abdomen in a patient with liver cancer and ascites can suggest peritonitis. Decreased breath sounds and rales in a lung cancer patient may indicate a developing pneumonia or empyema. Early determination of these factors can help guide initial management, including administration of fluids and empiric antibiotics, and inform clinical decision-making regarding the need for imaging.

Initial Diagnostic Workup

All patients with known malignancy and fever require a full infectious workup (Fig. 38.2). Though most patients with neutropenic fever may never have a diagnosed etiology, all patients presenting with febrile neutropenia should be treated as if they have true infection. After 48–72 h of continuous fever without positive culture data or response to antibiotics, alternative infectious etiologies should be considered including less common fungal, viral, and parasitic organisms.

Initial workup of an oncologic patient presenting with fever should include but is not limited to:

- Thorough history and physical examination to identify possible infectious sources including wounds, surgical sites, and indwelling central venous catheters and urinary catheters
- Complete blood count with differential, complete metabolic panel including hepatic function panels, other electrolytes, and serum lactate
- At least two sets of blood cultures from different sites, with consideration of obtaining a set from current indwelling central venous catheter or implantable port (if there is concern for infection involving the device)
- Microbiologic cultures from other sites including sputum, tracheostomy tubes, wounds, surgical incisions, CSF, and stool as applicable
- Urinalysis and urine cultures
- Chest radiography for the patient presenting with hypoxia, new oxygen requirements, or respiratory symptoms
- Brain imaging and lumbar puncture in a patient with altered mental status or neurological symptoms

Fig. 38.2 An approach to evaluation of the oncologic patient with neutropenic fever and/or suspected infection in the emergency department



Table 38.3 MASCC risk assessment [29]

Factors	Score
Burden of disease	
No or mild symptoms	5
Moderate symptoms	3
Severe symptoms	0
No hypotension	5
No chronic obstructive pulmonary disease	4
Solid tumor or hematologic malignancy without previous fungal infection	4
No dehydration	3
Outpatient status at onset of fever	3
Age of <60 years	2
Total score of 21 or higher means that the patient is considered low risk for complication	as

MASCC Multinational Association for Supportive Care in Cancer

 Nasopharyngeal swabs for patients with flu-like symptoms in the setting of community-acquired respiratory illnesses such as influenza or COVID-19

Ideally, patient assessment should be performed as soon as possible as delays in treatment can adversely affect outcomes. The goal of the emergency physician is not to identify a specific infectious diagnosis, but to maximize the chances of establishing clinical and microbiologic diagnoses later in the patient's clinical course that may affect treatment, including antibiotic choice and prognosis. In this same vein, the emergency physician must assess the patient's risk for infectious complications. Different risk stratification scoring systems have been proposed – the Multinational Association for Supportive Care in Cancer (MASCC) score is based on seven clinical factors derived and validated from prospective analyses (Table 38.3) [29].

While low-risk patients may be considered for outpatient therapy, high-risk patients will most likely require empiric intravenous antibiotics and hospitalization. Identifying a patient's risk for infectious complications is an important skill for the emergency physician as it ultimately determines the disposition of the patient and the level of care they will receive during their hospital course.

Treatment and Management

Guidelines from the American Society of Clinical Oncologic (ASCO) and Infectious Diseases Society of America (IDSA) for empiric treatment recommend the early use of broadspectrum antibiotics when treating febrile neutropenia [22, 28]. Stabilization of the sick patient is the critical goal for the emergency physician. Early aggressive fluid resuscitation should be considered in patients presenting with sepsis and/ or septic shock. Early blood products may also be required due to the marrow suppressing nature of cytotoxic chemotherapy. Obtaining central venous access, if not already present, and initiating vasopressor therapy can be crucial in preserving hemodynamics. Source control through removal of an infected central venous catheter or urinary catheter or drainage of an infected fluid collection (e.g., abscess) may be necessary. Early communication with the patient's oncologist or an on-call oncologist can help facilitate disposition and ensure proper next steps in management and follow-up.

Gram-positive or Gram-negative bacteria account for more than 90% of the causes for a first febrile episode in neutropenic patients [4]; therefore, empiric antibiotic regimens must be broad-spectrum and bactericidal and achieve therapeutic levels quickly (Fig. 38.3). Based on IDSA guidelines, patients should receive at least monotherapy with an antibiotic that empirically covers *P. aeruginosa* [28]. Reasonable choices therefore include an anti-pseudomonal fourth-generation cephalosporin (cefepime), β -lactam/ β -



Fig. 38.3 Empiric antibiotic therapy for suspected infection in highrisk oncologic patients

lactamase inhibitor (piperacillin-tazobactam), or carbapenem (meropenem). Vancomycin, or other agents targeted against resistant Gram-positives (linezolid, daptomycin), should be considered in patients who have suspected central venous catheter infection or severe skin and soft tissue infection. Septic patients may also require treatment with an aminoglycoside, especially when there is a history of resistant Gram-negative infection. Additional empiric coverage directed toward fungi and viruses can be considered based on the clinical context (e.g., previous fungal infection, previous isolation of multidrug-resistant organisms) and local antibiogram data [28].

Recent ASCO/IDSA guidelines have outlined a population of patients who may be safely treated for neutropenic fever outside of the hospital setting. Again, validated tools such as the MASCC index or the Clinical Index of Stable Febrile Neutropenia (CISNE) can aid decision-making [29, 30]. In stable, low-risk patients felt to be candidates for outpatient treatment, the current recommended antibiotic regimen includes fluoroquinolones active against *Pseudomonas* plus amoxicillin/clavulanate or clindamycin [22].

Case Study 1: An Oncologic Patient with Fever and Neutropenia

A 62-year-old woman with hypertension, COPD, and diffuse large B-cell lymphoma presents to the ED via private vehicle with fever. She first felt "hot" earlier this morning and found her temperature to be 102 °F. She has not had any other symptoms and feels well overall. She completed her first cycle of chemotherapy with R-CHOP (rituximab, cyclophos-

On arrival to the ED, the patient's vital signs are notable for a temperature of 102.5 °F, a heart rate of 98 beats per minute, and a blood pressure of 88/60 mmHg. A head-to-toe physical examination finds a well-appearing female with clear lung sounds, a non-tender abdomen, and good pulses in all her extremities. Initial laboratory assessment demonstrates a hemoglobin of 12.1 mg/dL and a white blood cell count of 1700 cells/µL. The laboratory calls the emergency physician with a critical lab result from the white blood cell differential - the absolute neutrophil count is 200 cells/ μ L. Given concern for neutropenic fever, the emergency physician obtains two sets of blood cultures and a urine culture and starts the patient empirically on intravenous cefepime, all within 1 h of her presentation to the ED. A chest radiograph shows no evidence of pneumonia and a urinalysis is within normal limits. The emergency physician determines that the patient is at high risk for infectious complications with a MASCC score of 15. She notifies the oncologist of the patient's neutropenic fever and recommends admission for further workup and observation. The oncologist concurs and is appreciative of the call.

During the hospitalization, the patient's blood and urine cultures show no growth. The oncologic team decides to administer granulocyte colony-stimulating factor. Three days later, her fever has resolved, and her absolute neutrophil count has increased to 500 cells/ μ L. She is discharged on oral ciprofloxacin and amoxicillin/clavulanate and is scheduled follow-up in 2 days with her oncologist. This case demonstrates the importance of early recognition of neutropenic fever, appropriate microbiological culture collection, and prompt initiation of empiric antibiotics, even in the absence of other localizing symptoms. Risk stratification tools such as the MASCC score and close communication with a patient's oncologist can help in clinical decisionmaking and securing timely outpatient follow-up once the patient has been discharged.

Special Considerations

Syndromic Approaches

Specific history and physical examination findings can help one to narrow their approach when diagnosing the cause of a fever in an oncologic patient. The following is a brief review of specific clinical syndromes and how one might approach them when it comes to infection.

Pulmonary syndromes usually manifest by increased work of breathing, hypoxia, and cough with or without sputum production. In neutropenic or stem cell transplant patients, pneumonia carries a high rate of mortality. Infiltrates
may not be prominent on routine chest radiography; thus computed tomography should be obtained in those who have clinical symptoms but a seemingly clear chest x-ray. Bacterial pathogens, such as S. pneumoniae, Staphylococcus aureus, and P. aeruginosa, predominate as causes of pneumonia in immunosuppressed oncologic patients [31, 32]. Given their frequent encounters with healthcare, resistant Gram-negative frequently organisms are also isolated, including **Stenotrophomonas** maltophilia, Acinetobacter spp., Klebsiella pneumoniae, etc. [31]. A high index of suspicion must also be maintained for fungi such as Aspergillus spp. and P. jirovecii [32]. In immunosuppressed patients, community respiratory viruses can cause pulmonary infiltrates on chest x-ray and result in significant morbidity. Influenza, respiratory syncytial virus, parainfluenza, and SARS-CoV-2 are among the many circulating viruses that can cause serious respiratory syndromes. Assessment should include routine labs and sputum culture, if the patient is making sputum. In critically ill patients who do not respond to empiric antibiotic therapy, bronchoalveolar lavage with microbiologic culture can help further guide coverage.

Skin and soft tissue infections have a broad differential in the immunosuppressed patient. The classic erythema, pain, warmth, and swelling of cellulitis may be less prominent than in immunocompetent patients. Clinicians should have a high index of suspicion for necrotizing infections, especially when patients have significant pain out of proportion to the exam. It is also important to recognize that in neutropenic or immunosuppressed patients, skin lesions are frequent manifestations of disseminated disease [5]. When evaluating skin infections, it is important not to overlook mucous membranes, which may help to differentiate cutaneous syndromes vs. systemic processes (such as disseminated viral processes). Vesicular, crusted, or shallow ulcerative lesions may suggest a viral process, such as disseminated zoster or herpes simplex virus. Nodular lesions may be suggestive of fungal or mycobacterial processes. Erythematous skin lesions with central necrosis can also signify a fungal process. Unfortunately, many non-infectious processes, including Sweet's syndrome, can mimic cutaneous infection. The diagnosis of these infections often relies on not only routine laboratory tests and blood cultures but frequently skin biopsy for pathology and special stains (e.g., Gram stain, GMS stain, Fite stain).

Gastrointestinal syndromes manifest as abdominal pain, nausea, vomiting, and diarrhea. Given their frequent exposure to healthcare and antibiotics, oncologic patients are at increased risk for *Clostridioides difficile* infection [33]. They are also at risk for a range of bacterial infections, including *E. coli, Salmonella, Shigella*, and *Campylobacter* [34]. Viral (CMV, norovirus) and parasitic (e.g., *Cryptosporidium, Strongyloides, Entamoeba*) causes should also be considered in immunosuppressed patients with unexplained diarrhea. Typhlitis is a life-threatening condition due to neutropenia resulting in necrotizing enterocolitis, predominantly affecting the cecum. This most often manifests as right lower quadrant abdominal pain, many times with associated bloody diarrhea [35]. This can be readily diagnosed on CT scan. Patients who have pain or difficulty swallowing should be evaluated for esophagitis, which may be due to Candida, CMV, or HSV infection. The approach to an immunosuppressed patient with abdominal pain often includes routine laboratory tests, including liver function testing, pancreatic enzymes, and stool studies (for C. difficile toxin or polymerase chain reaction, fecal leukocytes, and stool culture). When patients have significant pain, early CT scan should be considered. It should be noted that oncologic patients are also at risk for non-infectious causes of gastrointestinal illness, including tumor effects, medication effects, and, in HSCT patients, gastrointestinal graft-vs-host disease.

Central nervous system infections should be considered not only in patients with classic meningeal symptoms but also in those who are altered or obtunded. Emergency physicians should have a low threshold for evaluating patients for neurologic infections. Patients with symptoms concerning for focal neurologic deficits or meningoencephalitis should undergo imaging and early lumbar puncture as indicated. with appropriately recorded opening pressure. In addition to typical meningitis pathogens, such as S. pneumoniae, H. influenzae, and Neisseria meningitidis, immunosuppressed oncologic patients are at increased risk of infection due to pathogens such as Listeria monocytogenes, Cryptococcus neoformans, and herpesviruses (HSV/VZV) [35]. Those with a history of neurosurgical procedures have an increased risk for coagulase-negative staphylococci, S. aureus, and Gram-negative rods [36]. Early pathogen identification can improve time to life-saving therapies. Intracranial mass lesions can have many etiologies, including bacterial abscess, fungal infection, or Toxoplasma. The workup for these will often depend on imaging characteristics. Fungal sinusitis is a diagnosis that can mimic a neurologic syndrome based on severe headache and ocular nerve palsies. This is a diagnosis not to miss as the associated mortality is very high [37].

Healthcare-associated infections can arise from any number of interventions associated with cancer care. Central venous catheters and implantable ports are frequently used to provide chemotherapy, intravenous fluids, blood product transfusions, and other treatments. A central line-associated bloodstream infection (CLABSI) is defined by the Centers for Disease Control and Prevention as a primary bloodstream infection in a patient who has had a central venous catheter in place for more than 48 h with a bacterial or fungal pathogen isolated from one or more blood cultures unrelated to another source of infection [38]. While surveillance and clinical definitions may vary, a CLABSI can contribute significant morbidity and mortality, particularly in patients who have undergone HSCT [39, 40]. Coagulase-negative staphylococci, S. aureus, Enterococcus faecium, E. coli, and Candida are common causative organisms, with antibioticresistant organisms increasingly implicated [41, 42]. Fever, chills, and rigors may not always be present, particularly with infection due to coagulase-negative staphylococci. Erythema, induration, and purulence at the insertion site for the central venous catheter or overlying the pocket of an implantable port should raise suspicion that this device could be a potential source of infection. At least two sets of blood cultures should be obtained preferably from two separate peripheral sites; one set can be drawn from the central venous catheter or implantable port when necessary. When clinical suspicion for CLABSI or implantable port infection is high, device removal is preferred [42, 43]; this decision should be made in consultation with the patient's oncologist. Presence of an indwelling urinary catheter in an oncologic patient presenting with fever, suprapubic tenderness, costovertebral angle tenderness, or urinary symptoms should raise concern for a catheter-associated urinary tract infection (CAUTI). Urinalysis and urine culture help establish the diagnosis and guide antibiotic therapy; catheter removal is strongly recommended. Avoidance of unnecessary urinary catheterization, including in the ED, should remain a major strategy to prevent CAUTI [44].

Case Study 2: A Thorough Examination Uncovers the Source of Fever in an III Oncologic Patient

A 73-year-old man with hypertension, coronary disease, and small-cell lung cancer presents to the ED via emergency medical services for altered mentation and hypotension. Approximately 12 h prior to admission, the patient developed fever at home, and his mental status subsequently declined throughout the day. His wife eventually called EMS and on arrival he is obtunded and hypotensive.

In the ED, the patient is febrile to 101.3 °F with a heart rate of 120 beats per minute and a blood pressure of 73/29 mmHg. A head-to-toe physical examination is performed. He is obtunded but has no focal deficits and is protecting his airway. His heart and lung exam are significant for crackles in the lung bases. Abdominal examination is benign. On exposure of the chest during a complete skin examination, mild erythema and induration are visualized overlying the site of his implantable port. Initial laboratory assessment demonstrates anemia (hemoglobin 9.7 mg/dL), thrombocytopenia (platelet count 129 × 10°/L), and a white blood cell count of 6000 cells/µL. Two sets of blood cultures are obtained, and the patient is started empirically on intravenous vancomycin. A central venous catheter is placed and vasoactive agents are started. It is noted the patient recently completed cycle 3 of cisplatin and etoposide for his cancer. He has not had any recent procedures.

The patient is admitted to the oncologic ICU. Twelve hours into his stay, blood cultures are reported as positive for Gram-positive cocci in clusters. The following day, his implantable port is removed. The organisms in blood cultures are identified as methicillin-resistant Staphylococcus aureus. Two days later, he is improved and able to transition out of the ICU. This case demonstrates the importance of a thorough physical examination in identifying potential sources of fever in an oncologic patient, including evaluation of vascular access devices commonly used for cancer treatment.

Future Issues and Considerations

In the United States, there are over 14 million people who have been treated for cancer sometime during their life [44]. While outpatient therapy is the mainstay of cancer management, the ED is the first destination for many patients with acute complications. National data suggests that patients with cancer make up approximately 3% of all ED visits with an admission rate that is much higher than that of the general ED population; future studies should focus on improving emergency care, especially with regard to febrile neutropenia [45]. As stated before, guidelines recommend early administration of empiric antibiotics for patients who present with neutropenic fever. However, in the setting of ED overcrowding and prolonged boarding times, significant delays in initiating treatment have been observed, with some studies reporting a median time to initial antibiotics ranging from 102 to 300 min [46]. Understanding the barriers that prevent prompt treatment and care can provide systemic targets for improving overall patient flow in the ED. Another consideration is the continued efforts to develop risk stratification methods to identify patients with the highest risk of serious infection and need for immediate assessment. As protocols and systems have developed to improve intervention times of other emergent presentations such as myocardial infarction and stroke, similar systems can be developed to reduce the time to initial antibiotic treatment. However, these strategies must be based on adequately powered and valid studies with convincing evidence of positive clinical effects and outcomes [45].

As physicians wage a never-ending arms race against evolving multidrug-resistant bacteria, it is important to consider the potential ramifications of continuing broadspectrum antibiotics when it comes to antibiotic stewardship. While inadequate antibiotic treatment can lead to worse clinical outcomes in oncologic patients, continued use of broad-spectrum therapy when not indicated promotes selection of antibiotic-resistant bacteria. As the use of antibiotics has evolved over time, causative organisms associated with febrile neutropenia have shifted from Gram-negative bacteria in the 1960s and 1970s to Gram-positive bacteria in the 1980s and now to antibiotic-resistant Gram-negative and Gram-positive bacteria in recent years [47]. Multidrugresistant organisms including K. pneumoniae, P. aeruginosa, and MRSA are associated with considerable mortality in patients with and without neutropenia, although data regarding oncologic patients is limited [48]. Risk assessment for multidrug-resistant infection should be considered for every oncologic patient, particularly those with a history of frequent hospital admissions. While empiric broad-spectrum antibiotic therapy is warranted in the management of undifferentiated infection in a cancer patient, emergency physicians have an important role to play in promoting antibiotic stewardship through responsible prescribing. Selecting the right drug (targeting the most likely organisms with as narrow a spectrum as allowable; using hospital antibiograms to guide antibiotic selection based on trends in local antibiotic resistance), administering that drug at the right dose (e.g., accounting for hepatic and/or renal clearance) for the right duration (shortest time needed to appropriately treat the infection), enabling opportunities to de-escalate antibiotic therapy based on microbiologic cultures, and making sure a drug is prescribed for the right diagnosis in the ED (avoiding antibiotic use when a viral infection is suspected) constitute the 5 Ds of antibiotic stewardship in emergency medicine [41].

Summary and Key Points

The oncologic patient presents a diagnostic challenge to the emergency physician due to their immunocompromised state, lack of typical inflammatory signs and symptoms, and potential for atypical clinical presentations. Fever in a neutropenic patient is an oncologic emergency that necessitates immediate, thorough assessment and early, empiric treatment. When evaluating these patients, the following key points should be remembered:

- Neutropenic fever is defined as a temperature ≥38.3 °C once or ≥38.0 °C sustained over an hour in the setting of absolute neutrophil count (ANC) <500 cells/µL or an ANC <1000 with predicted nadir <500 cells/µL.
- Malignancies can also cause fever, but all fevers must be treated as infectious until proven otherwise.
- Understanding the rationale for antimicrobial prophylaxis in a cancer patient can help narrow the diagnostic possibilities when ill oncologic patients present to the ED.
- A thorough history and physical examination can help uncover subtle signs and symptoms of infection.

- A broad workup should be performed to help maximize the chances of establishing a clinical and microbiological diagnosis.
- Blood cultures should be obtained and empiric broadspectrum antibiotics started within 1 h of presentation.
- Validated tools such as the MASCC index or the Clinical Index of Stable Febrile Neutropenia (CISNE) can help with risk stratification and determination of disposition.
- Future studies should focus on improving time to antibiotics, risk stratification, and addressing the growing challenge of multidrug-resistant bacteria.

References

- Kuderer NM, Dale DC, Crawford J, Cosler LE, Lyman GH. Mortality, morbidity, and cost associated with febrile neutropenia in adult cancer patients. Cancer. 2006;106(10):2258–66.
- White L, Ybarra M. Neutropenic fever. Emerg Med Clin North Am. 2014;32(3):549–61.
- Bow E, Wingard JR. Overview of neutropenic fever syndromes. In: Post TW, editor. UpToDate. Waltham MA: UpToDate; 2020.
- Marchetti O, Tissot F, Calandra T. Infections in the cancer patient. In: Cohen J, Powderly WG, Opal SM, editors. Infectious diseases. 4th ed. Philadelphia: Elsevier Ltd; 2017. p. 723–38.
- Burke VE, Lopez FA. Approach to skin and soft tissue infections in non-HIV immunocompromised hosts. Curr Opin Infect Dis. 2017;30(4):354–63.
- 6. Tomblyn M, Chiller T, Einsele H, Gress R, Sepkowitz K, Storek J, et al.; Center for International Blood and Marrow Research; National Marrow Donor program; European Blood and Marrow Transplant Group; American Society of Blood and Marrow Transplantation; Canadian Blood and Marrow Transplant Group; Infectious Diseases Society of America; Society for Healthcare Epidemiology of America; Association of Medical Microbiology and Infectious Disease Canada; Centers for Disease Control and Prevention. Guidelines for preventing infectious complications among hematopoietic cell transplantation recipients: a global perspective. Biol Blood Marrow Transplant. 2009;15(10):1143–238.
- Centers for Disease Control and Prevention; Infectious Disease Society of America; American Society of Blood and Marrow Transplantation. Guidelines for preventing opportunistic infections among hematopoietic stem cell transplant recipients. MMWR Recomm Rep. 2000;49(RR-10):1–125, CE1–7. Erratum in: MMWR Recomm Rep. 2004;53(19):396.
- Mikulska M, Averbuch D, Tissot F, Cordonnier C, Akova M, Calandra T, et al.; European Conference on Infections in Leukemia (ECIL). Fluoroquinolone prophylaxis in haematological cancer patients with neutropenia: ECIL critical appraisal of previous guidelines. J Infect. 2018;76(1):20–37.
- Bucaneve G, Micozzi A, Menichetti F, Martino P, Dionisi MS, Martinelli G, et al.; Gruppo Italiano Malattie Ematologiche dell'Adulto (GIMEMA) Infection Program. Levofloxacin to prevent bacterial infection in patients with cancer and neutropenia. N Engl J Med. 2005;353(10):977–87.
- Boeckh M, Nichols WG. The impact of cytomegalovirus serostatus of donor and recipient before hematopoietic stem cell transplantation in the era of antiviral prophylaxis and preemptive therapy. Blood. 2004;103(6):2003–8.

- Pande A, Dubberke ER. Cytomegalovirus infections of the stem cell transplant recipient and hematologic malignancy patient. Infect Dis Clin N Am. 2019;33(2):485–500.
- Goodrich JM, Mori M, Gleaves CA, Du Mond C, Cays M, Ebeling DF, et al. Early treatment with ganciclovir to prevent cytomegalovirus disease after allogeneic bone marrow transplantation. N Engl J Med. 1991;325(23):1601–7.
- Marty FM, Ljungman P, Chemaly RF, Maertens J, Dadwal SS, Duarte RF, et al. Letermovir prophylaxis for cytomegalovirus in hematopoietic-cell transplantation. N Engl J Med. 2017;377(25):2433–44.
- De Castro N, Neuville S, Sarfati C, Ribaud P, Derouin F, Gluckman E, et al. Occurrence of Pneumocystis jirovecii pneumonia after allogeneic stem cell transplantation: a 6-year retrospective study. Bone Marrow Transplant. 2005;36(10):879–83.
- Marr KA, Seidel K, Slavin MA, Bowden RA, Schoch HG, Flowers ME, et al. Prolonged fluconazole prophylaxis is associated with persistent protection against candidiasis-related death in allogeneic marrow transplant recipients: long-term follow-up of a randomized, placebo-controlled trial. Blood. 2000;96(6):2055–61.
- Erard V, Guthrie KA, Varley C, Heugel J, Wald A, Flowers ME, et al. One-year acyclovir prophylaxis for preventing varicella-zoster virus disease after hematopoietic cell transplantation: no evidence of rebound varicella-zoster virus disease after drug discontinuation. Blood. 2007;110(8):3071–7.
- Boeckh M, Nichols WG, Chemaly RF, Papanicolaou GA, Wingard JR, Xie H, et al. Valganciclovir for the prevention of complications of late cytomegalovirus infection after allogeneic hematopoietic cell transplantation: a randomized trial. Ann Intern Med. 2015;162(1):1–10.
- Ullmann AJ, Lipton JH, Vesole DH, Chandrasekar P, Langston A, Tarantolo SR, et al. Posaconazole or fluconazole for prophylaxis in severe graft-versus-host disease. N Engl J Med. 2007;356(4):335–47.
- Taplitz RA, Kennedy EB, Bow EJ, Crews J, Gleason C, Hawley DK, et al. Antimicrobial prophylaxis for adult patients with cancerrelated immunosuppression: ASCO and IDSA clinical practice guideline update. J Clin Oncol. 2018;36(30):3043–54.
- 20. Gafter-Gvili A, Fraser A, Paul M, van de Wetering M, Kremer L, Leibovici L. Antibiotic prophylaxis for bacterial infections in afebrile neutropenic patients following chemotherapy. Cochrane Database Syst Rev. 2005;(4):CD004386. Update in: Cochrane Database Syst Rev. 2012;(1):CD004386.
- Leibovici L, Paul M, Cullen M, Bucaneve G, Gafter-Gvili A, Fraser A, Kern WV. Antibiotic prophylaxis in neutropenic patients: new evidence, practical decisions. Cancer. 2006;107(8):1743–51.
- 22. Taplitz RA, Kennedy EB, Bow EJ, Crews J, Gleason C, Hawley DK, et al. Outpatient management of fever and neutropenia in adults treated for malignancy: American Society of Clinical Oncology and Infectious Diseases Society of America Clinical Practice Guideline Update. J Clin Oncol. 2018;36(14):1443–53.
- 23. del Castillo M, Romero FA, Argüello E, Kyi C, Postow MA, Redelman-Sidi G. The spectrum of serious infections among patients receiving immune checkpoint blockade for the treatment of melanoma. Clin Infect Dis. 2016;63(11):1490–3.
- 24. Toussaint E, Bahel-Ball E, Vekemans M, Georgala A, Al-Hakak L, Paesmans M, Aoun M. Causes of fever in cancer patients (prospective study over 477 episodes). Support Care Cancer. 2006;14(7):763–9.
- 25. Foggo V, Cavenagh J. Malignant causes of fever of unknown origin. Clin Med (Lond). 2015;15(3):292–4.
- Kurzrock R. The role of cytokines in cancer-related fatigue. Cancer. 2001;92(6 Suppl):1684–8.
- Kudlowitz D. Neoplastic fever: pathophysiology, clinical features, and diagnostic assessment. Clinical Correlations. NYU Langone Online Journal of Medicine. 13 Aug 2014. https://www.clinical-

correlations.org/2014/08/13/neoplastic-fever-pathophysiologyclinical-features-and-diagnostic-assessment/. Accessed 29 June 2020.

- 28. Freifeld AG, Bow EJ, Sepkowitz KA, Boeckh MJ, Ito JI, Mullen CA, et al.; Infectious Diseases Society of America. Clinical practice guideline for the use of antimicrobial agents in neutropenic patients with cancer: 2010 update by the infectious diseases society of America. Clin Infect Dis. 2011;52(4):e56–93.
- 29. Klastersky J, Paesmans M, Rubenstein EB, Boyer M, Elting L, Feld R, et al. The Multinational Association for Supportive Care in Cancer risk index: a multinational scoring system for identifying low-risk febrile neutropenic cancer patients. J Clin Oncol. 2000;18(16):3038–51.
- 30. Carmona-Bayonas A, Jiménez-Fonseca P, Virizuela Echaburu J, Antonio M, Font C, Biosca M, et al. Prediction of serious complications in patients with seemingly stable febrile neutropenia: validation of the clinical index of stable febrile neutropenia in a prospective cohort of patients from the FINITE study. J Clin Oncol. 2015;33(5):465–71.
- Evans SE, Ost DE. Pneumonia in the neutropenic cancer patient. Curr Opin Pulm Med. 2015;21(3):260–71.
- 32. Rañó A, Agustí C, Jimenez P, Angrill J, Benito N, Danés C, et al. Pulmonary infiltrates in non-HIV immunocompromised patients: a diagnostic approach using non-invasive and bronchoscopic procedures. Thorax. 2001;56(5):379–87.
- Gorschlüter M, Glasmacher A, Hahn C, Schakowski F, Ziske C, Molitor E, et al. Clostridium difficile infection in patients with neutropenia. Clin Infect Dis. 2001;33(6):786–91.
- Krones E, Högenauer C. Diarrhea in the immunocompromised patient. Gastroenterol Clin North Am. 2012;41(3):677–701.
- Cantwell L, Perkins J. Infectious disease emergencies in oncology patients. Emerg Med Clin North Am. 2018;36(4):795–810.
- Safdieh JE, Mead PA, Sepkowitz KA, Kiehn TE, Abrey LE. Bacterial and fungal meningitis in patients with cancer. Neurology. 2008;70(12):943–7.
- Monroe MM, McLean M, Sautter N, Wax MK, Andersen PE, Smith TL, Gross ND. Invasive fungal rhinosinusitis: a 15-year experience with 29 patients. Laryngoscope. 2013;123(7):1583–7.
- 38. Centers for Disease Control and Prevention/National Healthcare Safety Network. Bloodstream Infection Event (Central Line-Associated Bloodstream Infections and non-Central Line Associated Blood Stream Infection). In: National Healthcare Safety Network (NHSN) Patient Safety Component Manual. Jan 2020. https://www.cdc.gov/nhsn/pdfs/pscmanual/pcsmanual_ current.pdf.
- 39. Lukenbill J, Rybicki L, Sekeres MA, Zaman MO, Copelan A, Haddad H, et al. Defining incidence, risk factors, and impact on survival of central line-associated blood stream infections following hematopoietic cell transplantation in acute myeloid leukemia and myelodysplastic syndrome. Biol Blood Marrow Transplant. 2013;19(5):720–4.
- 40. McDonald MK, Culos KA, Gatwood KS, Prow C, Chen H, Savani BN, et al. Defining incidence and risk factors for catheterassociated bloodstream infections in an outpatient adult hematopoietic cell transplantation program. Biol Blood Marrow Transplant. 2018;24(10):2081–7.
- 41. See I, Freifeld AG, Magill SS. Causative organisms and associated antimicrobial resistance in healthcare-associated, central line-associated bloodstream infections from oncology settings, 2009-2012. Clin Infect Dis. 2016;62(10):1203–9.
- 42. Chaftari AM, Hachem R, Jiang Y, Shah P, Hussain A, Hamal ZA, et al. Changing epidemiology of catheter-related bloodstream infections in cancer patients. Infect Control Hosp Epidemiol. 2018;39(6):727–9.
- 43. Lo E, Nicolle LE, Coffin SE, Gould C, Maragakis LL, Meddings J, et al. Strategies to prevent catheter-associated urinary tract infec-

tions in acute care hospitals: 2014 update. Infect Control Hosp Epidemiol. 2014;35(Suppl 2):S32–47.

- 44. DeSantis CE, Lin CC, Mariotto AB, Siegel RL, Stein KD, Kramer JL, et al. Cancer treatment and survivorship statistics, 2014. CA Cancer J Clin. 2014;64(4):252–71.
- 45. Brown J, Grudzen C, Kyriacou DN, Obermeyer Z, Quest T, Rivera D, et al. The emergency care of patients with cancer: setting the research agenda. Ann Emerg Med. 2016;68(6):706–11.
- 46. Kyriacou DN, Jovanovic B, Frankfurt O. Timing of initial antibiotic treatment for febrile neutropenia in the emergency department: the

need for evidence-based guidelines. J Natl Compr Cancer Netw. 2014;12(11):1569-73.

- Zinner SH. Changing epidemiology of infections in patients with neutropenia and cancer: emphasis on gram-positive and resistant bacteria. Clin Infect Dis. 1999;29(3):490–4.
- 48. Perdikouri EIA, Arvaniti K, Lathyris D, Apostolidou Kiouti F, Siskou E, Haidich AB, Papandreou C. Infections due to multidrug-resistant bacteria in oncological patients: insights from a five-year epidemiological and clinical analysis. Microorganisms. 2019;7(9):277.

Febrile Neutropenia

Christopher J. Coyne and Rahul V. Nene

Case Study

A 54-year-old female with a history of breast cancer presents to the ED with a low-grade fever (38.3 °C) and myalgias. She denies cough, shortness of breath, chest pain, abdominal pain, dysuria, sore throat, or mouth sores. Her physical exam is non-focal and does not reveal a potential source of infection. The patient has a past medical history of essential hypertension, though is otherwise healthy and exercises every day. Her labs are significant for an absolute neutrophil count of 300 cells/mm³ and an absolute monocyte count of 600 cells/mm³. All additional labs including a complete metabolic panel, lactate, and urinalysis are within normal limits. She receives a chest x-ray that is read as negative for acute cardiopulmonary disease. Blood cultures, a urine culture, and a respiratory viral panel are sent and pending.

Given the reassuring clinical picture, the emergency physician applies the Clinical Index of Stable Febrile Neutropenia (CISNE) score to assist with risk stratification. The patient is found to have a CISNE score of zero and is therefore considered low risk of developing severe disease. She is placed in observation status and receives an initial dose of amoxicillin/clavulanate and ciprofloxacin. The patient remains stable for several hours while in the ED. After a discussion between the emergency physician and the patient's primary oncologist, the patient is cleared for discharge home with a plan for expedited follow-up. She will continue oral antibiotic therapy and she will return if any new or concerning issues arise.

C. J. Coyne (🖂)

R. V. Nene Department of Emergency Medicine, University of California, San Diego, San Diego, CA, USA

Introduction

Infectious complications related to cancer and cytotoxic chemotherapy remain common. A particular disease process that has the potential for significant morbidity and mortality is febrile neutropenia. Appropriate identification and management of this condition is critical given that major complications (e.g., end-organ damage, need for mechanical ventilation) occur in up to 30% of patients and the reported mortality may approach 11% [1-3]. Unfortunately, the classic signs and symptoms of infection in patients with neutropenia are often absent or muted, with fever being the only presenting symptom in many cases [4]. This makes it difficult for the treating clinician to diagnose the underlying cause and provide directed therapy against potential infectious agents. Almost universally, patients are treated for potentially serious and life-threatening bacterial infections. Many patients, however, may suffer from viral or fungal pathogens, while others may experience fever due to underlying malignancy and/or therapeutic agents. Due to this diagnostic dilemma, the ultimate disposition of these patients becomes quite challenging. Which patients can safely be discharged and which require hospitalization?

In this chapter, we will address the most recent literature and recommendations relating to the diagnosis and treatment of febrile neutropenia in the acute setting. We will discuss the current best-practice guidelines in the identification of a low-risk patient cohort, who may be safely treated in the outpatient setting. Finally, we will discuss future considerations in febrile neutropenia diagnosis and risk stratification.

Definition

Febrile neutropenia is a vitally important complication of chemotherapy. Neutrophils are an essential component of the immune response, especially against bacteria and fungi, and febrile neutropenia may be the only sign of a serious infection in an immunocompromised patient [1, 5]. Neutropenia

Check for updates

Department of Emergency Medicine, Department of Radiation Medicine and Applied Sciences, University of California, San Diego, San Diego, CA, USA e-mail: cjcoyne@ucsd.edu

is typically categorized as mild, with an absolute neutrophil count (ANC) 1000–1500 cell/ μ L; moderate, with an ANC 500–1000 cells/ μ L; and severe, with an ANC <500 cells/ μ L [6]. Per the American Society of Clinical Oncologic (ASCO) and Infectious Diseases Society of America (IDSA), febrile neutropenia is defined as a single oral temperature (either obtained in a healthcare setting or self-reported) of \geq 38.3 °C (101°F) or a sustained temperature of \geq 38.0 °C (100.4°F) for 1 h, with an ANC <500 or an ANC expected to decrease to <500 in the next 48 h [5, 7].

Pathophysiology

Many cytotoxic chemotherapy regimens work by targeting rapidly dividing cancer cells, but in the process also damage the patient's normal cells, including those that make up the immune system and mucosal linings. Not only does this leave the patient immunocompromised, but the breakdown of mucosa anywhere along the gastrointestinal tract allows for seeding of the bloodstream with endogenous flora; this is thought to cause the majority of febrile neutropenia cases [4]. Patients may also be more susceptible to infection because of the underlying cancer. Hematologic malignancies by definition have a defect in some aspect of the immune system. Solid tumors can also cause obstruction to the hepatobiliary, gastrointestinal, bronchial, and renal tracts, which can serve as a nidus for infection. Surgery and the presence of implantable venous access devices also pose a risk of infection.

Neutropenia normally occurs 5-10 days after chemotherapy [8]. Patients with solid tumors undergoing cytotoxic chemotherapy typically still have healthy bone marrow, and neutropenia generally lasts less than 7 days. In contrast, patients with hematologic malignancies undergoing chemotherapy and/or hematopoietic stem cell transplantation (HSCT) may have extended periods of neutropenia lasting more than 14 days [9, 10]. It is estimated that 5-30% of patients receiving chemotherapy for solid tumors will develop febrile neutropenia, while the rate of this complication is >80% for patients with leukemia or those undergoing HSCT 5. Other factors that increase the risk of febrile neutropenia include metastatic cancer, mucositis, chronic obstructive pulmonary disease, and advanced age [11]. The cytotoxic chemotherapies associated with the highest risk of neutropenia include anthracyclines, taxanes, platinums, ifosfamide, cyclophosphamide, etoposide, and cytarabine [11, 12].

An infectious source is identified in only 40-50% of febrile neutropenia cases, with bacteremia identified in

10-25% [7, 13]. An additional 20-25% of patients have an identifiable site of infection (e.g., pneumonia, cellulitis), but no identifiable causative organism [10]. Less than 5% of cases have a non-infectious cause of fever, such as drug fever or tumor fever; as such, this should be a diagnosis of exclusion [14]. It is suspected 80% of infections arise from endogenous flora. Interestingly, over time there have been changes in the most common identified pathogens. Historically until the 1980s, gram-negative bacilli predominated, particularly Pseudomonas aeruginosa, Escherichia coli, Klebsiella species, and *Enterobacter* species [10, 15]. Since then, grampositive cocci have become more common, particularly Staphylococcus epidermidis, and, to a lesser extent, Staphylococcus aureus, and streptococci [16, 17]. It is suspected the cause of this transition is the increasing prevalence of implantable venous access devices, as well as the use of prophylactic fluoroquinolone antibiotics, which are primarily active against gram-negative species. More recently however, there has been a trend back toward gramnegative bacteremia, particularly multidrug-resistant organisms [18].

Fungal pathogens can also cause neutropenic fever, though they are rarely associated with the first episode [10]. The risk of fungal infection increases with the duration and severity of the neutropenia, particularly with profound neutropenia lasting more than 10–15 days [19]. Most commonly, *Candida* species were identified in central venous catheterassociated infections and gastrointestinal tract colonization, while *Aspergillus* was identified more commonly with respiratory tract infections [20]. Severely immunocompromised patients are also at risk for mucormycosis, invasive *Fusarium* infection, and new infection or reactivation of endemic fungi such as *Coccidioides*, *Histoplasma*, and *Blastomyces* [10, 20]. Other opportunistic infections should also be considered, including tuberculosis, *Pneumocystis jirovecii*, and *Mycobacterium avium complex*.

Viral infections are also a common cause of febrile neutropenia. Community-acquired respiratory viruses are of particular concern and have been documented with increasing frequency in neutropenic patients, particularly seasonal influenza, respiratory syncytial virus, parainfluenza, adenovirus, rhinoviruses, human metapneumovirus, and coronaviruses, including the novel corona virus, COVID-19 [21, 22]. Reactivation of human herpesviruses, particularly herpes simplex virus-1 and 2, and herpes zoster, is also very common, however typically only in those with hematologic malignancies and prolonged neutropenia. As a result, it is standard practice to start empiric antiviral prophylaxis on most patients with hematologic malignancies prior to the commencement of chemotherapy [7, 23].

Assessment

A patient with febrile neutropenia may present with vague nonspecific complaints [24]. Due to the muted inflammatory response in an immunocompromised patient, fever may often be the only presenting sign or symptom. The assessment of temperature can be particularly tricky and is often the turning point between discharging a patient home or starting broad-spectrum antibiotics in the hopes of not missing life-threatening illness. An oral temperature is the preferred method, though care should be taken in those with significant oral mucositis, which may also falsely elevate the temperature [25]. Rectal thermometry is not recommended in neutropenic patients due to the potential risk of mucosal injury leading to bacteremia.

A thorough history should be obtained to evaluate the risk of serious complications and help determine the potential site of infection [7]. Important factors to assess include medical comorbidities, a thorough review of systems, chemotherapeutic regimen and timing of last cycle, use of antibiotic prophylaxis or recent antibiotic treatment, and history of prior infections or colonization with multidrug-resistant organisms. A careful physical examination should be performed, with special attention to the skin, biopsy sites, catheter sites, oropharynx, lungs, abdomen, genitals and perianal area, and decubitus sites.

Patients with febrile neutropenia require a broad diagnostic workup, consisting of a complete blood cell count with differential, a complete metabolic panel with liver function tests, and an evaluation of all possible sources of infection. This includes, but is not limited to, urinalysis, urine culture, two blood cultures obtained from peripheral sites and additional blood cultures from each central venous catheter, sputum culture for respiratory symptoms, and stool sample for gastrointestinal symptoms. Consider lumbar puncture and cerebrospinal fluid (CSF) cell count, culture, and PCR testing if there is a high index of suspicion for meningitis or encephalitis. Consider fungal blood cultures, fungal serum markers, and a beta-D-glucan assay in patients with persistent neutropenic fever or a history of prior invasive fungal infection. Have a low threshold for respiratory viral testing, especially during influenza season or other viral epidemic. However, caution should be exercised when interpreting any of these tests in a neutropenic patient; the absence of typical laboratory findings does not exclude infection in the immunocompromised patient.

Patients with respiratory symptoms should have a chest radiograph, with a low threshold to proceed to chest computed tomography (CT) given the high prevalence of occult pneumonia [26]. Additional advanced imaging can be obtained according to suggestive symptoms (e.g., head, sinuses). Of particular note is CT of the abdomen/pelvis,

Table 39.1 Diagnostic evaluation of febrile neutropenia

Should obtain for all patients
CBC with diff
CMP
Peripheral blood cultures $\times 2$
Urinalysis and urine culture
Chest x-ray
Strongly consider for all patients
Viral respiratory pathogen panel
Patient-specific additional testing
Additional blood cultures from each CVC
Sputum culture
Stool culture
Fungal cultures
Beta-D-glucan
CSF cell count and culture
CT chest/abdomen/pelvis

CBC complete blood count, *CMP* complete metabolic panel, *CVC* central venous catheter, *CSF* cerebrospinal fluid

with contrast when possible, which should be obtained for patients with any abdominal complaints, especially if there is a concern for neutropenic enterocolitis (Table 39.1).

Management

Classically, the management of febrile neutropenia has included broad-spectrum antibiotics and hospital admission. This approach minimizes morbidity and mortality by providing aggressive empirical care prior to obtaining microbiological isolates. Since the 1960s, this approach has been supported by a large body of literature in the fields of oncologic, infectious diseases, and emergency medicine [27, 28]. Additionally, hospitalization was classically continued until fevers resolved and neutrophil counts recovered. Subsequent literature, however, suggested that a more nuanced, less aggressive approach may be appropriate for a subset of lowrisk patients [29, 30]. This section will first cover febrile neutropenia risk stratification, followed by a discussion on empiric antibiotic therapy and disposition considerations.

Risk Stratification

The risk of severe disease among patients with febrile neutropenia depends on several factors, including the cancer type, medical comorbidities, and a variety of findings on history and physical exam. Patients with a prior history of opportunistic infections, those with anticipated prolonged neutropenia (>7 days less than 100 ANC), or those who have received more intense cytotoxic chemotherapy are known to be of higher risk. Patients with risk of prolonged neutropenia include those who have received several courses of **Table 39.2** MASCC Risk Index Score ≥ 21 indicates that the patient is low risk for complications

Characteristic	Weight
Burden of illness with no or mild symptoms	5
Burden of illness with moderate symptoms	3
No hypotension (systolic blood pressure >90 mm Hg)	5
No chronic obstructive pulmonary disease	4
Solid tumor or no previous fungal infection	4
No dehydration requiring parental fluids	3
Outpatient status at onset of fever	3
Age <60 years	2

MASCC Multinational Association for Supportive Care in Cancer

myelosuppressive chemotherapy, allogeneic stem cell recipients, patients with bone marrow metastases, and those who have received radiation to the pelvis or long bones [31]. Additionally, those patients who receive alemtuzumab are known to be at particular high risk. Patients with a lower baseline functional status (Eastern Cooperative Oncologic Group Performance Scale [ECOG PS] \geq 2) are more likely to develop severe disease. Other indicators of high-risk febrile neutropenia include hepatic or renal insufficiency, uncontrolled or progressive cancer, grade 3–4 mucositis, or those with more life-threatening infectious sources (pneumonia, meningitis, encephalitis, etc.) [31].

One of the most widely accepted tools to aid clinicians in the risk stratification of patients with febrile neutropenia is the Multinational Association for Supportive Care in Cancer (MASCC) score. First published in 2000 by Klastersky et al., the MASCC score synthesizes of variety of clinical and historical data to produce a neutropenic fever risk score (Table 39.2) [30]. Points are assigned based on the lack of high-risk features with a score ≥ 21 considered as low risk and potentially appropriate for outpatient management (maximum score of 26). Of note, this score was only validated in the adult population and should not be used to risk stratify children.

A more recent risk stratification tool, the Clinical Index of Stable Febrile Neutropenia (CISNE) score, was validated in 2015 by Carmona-Bayonas et al. and specifically targets those patients with solid tumor malignancies (Table 39.3) [2]. This score utilizes similar historical risk factors as the MASCC score but eliminates symptom burden and adds certain laboratory values (absolute monocyte count, presence of hyperglycemia) as well as functional status (ECOG PS). In studies that compared the CISNE and MASCC scores, the CISNE score appeared to have a greater sensitivity in the identification of a low-risk cohort, with fewer patients misclassified as low risk [32, 33]. The most recent American Society of Clinical Oncologic (ASCO) and Infectious Diseases Society of America (IDSA) Joint guideline on the safe discharge of low-risk patients with febrile neutropenia supported the application of the CISNE score to patients with solid tumor malignancy who have received mild- to

Points
2
2
1
1
1
1

ECOG PS Eastern Cooperative Oncologic Group Performance Scale

moderate-intensity chemotherapy. Studies evaluating these **Table 39.4** ASCO/IDSA safe discharge criteria for patients with febrile neutropenia

	Residence ≤ 1 h or ≤ 30 miles (48 km) from clinic or hospital
	Patient's primary care physician or oncologist agrees to outpatient management
	Able to comply with logistic requirements, including frequent clinic visits
	Family member or caregiver at home 24 h/d
	Access to a telephone and transportation 24 h/d
	No history of noncompliance with treatment protocols
Tl	he following additional measures are recommended:
	Frequent evaluation for at least 3 days in clinic or at home
	Daily or frequent telephone contact to verify (by home thermometry) that fever resolves
	Monitoring of ANC and platelet count for myeloid reconstitution
	Frequent return visits to clinic

risk stratification tools in the ED have similarly supported utilizing the CISNE score to identify low-risk febrile neutropenia patients with solid malignancy who may be suitable for outpatient management [34].

With regard to febrile neutropenia risk stratification in the pediatric population, there is no universally accepted risk stratification tool. To date, there have been at least 27 attempts to derive a validated score to appropriately categorize pediatric patients as low or high risk; however, none of these scores have achieved adequate sensitivity or specificity [35–38].

Finally, there are several social considerations that must be addressed prior to safely discharging low-risk febrile neutropenic patients. The ASCO/IDSA safe-discharge criteria are listed in Table 39.4 [1].

Antibiotic Therapy

It is recommended that all patients presenting with febrile neutropenia rapidly receive broad-spectrum empiric antibiotics. The joint ASCO/IDSA guidelines state that all patients with febrile neutropenia should receive appropriate antibiotic therapy within 1 h of triage [1]. Perron et al. demonstrated that early antibiotics significantly decreased hospital length of stay for patients with febrile neutropenia who received early antibiotics. Rosa et al. found there to be a decrease in 28-day mortality for patients who received early antibiotic therapy (within 30 min), with an increase in 28-day mortality by 18% for each hour that antibiotics were delayed [39, 40]. A more recent, well-powered study by Daniels et al., however, demonstrated that only more significant delays in antibiotic administration (>3 h) were associated with increased length of stay and 30-day mortality [41].

Several factors must be considered when choosing the initial antibiotic course including previous culture data, the presence of indwelling catheters, site of infection, organ dysfunction, drug allergy, previous antibiotic therapy, and local antibiograms/sensitivities. Choosing the correct empiric antibiotic is likely the most important modifiable factor affecting the morbidity and mortality of a patient with febrile neutropenia [42, 43]. Previous literature has demonstrated that inappropriate empiric antibiotics may lead to a significant increase in mortality, especially in cases of Pseudomonas aeruginosa [44]. Consensus guidelines have been published by the National Comprehensive Cancer Network (NCCN) to guide clinicians in the initial selection of antibiotics for patients with febrile neutropenia (Table 39.5) [31]. Note, however, that one should synthesize all available information (previous cultures, specific risk factors, etc.) before choosing an initial antibiotic.

In an era of increasing antibiotic resistance, there are emerging data that suggest alternating first-line therapies may be beneficial to decrease antibiotic resistance. This strategy has been shown to be effective in increasing antibiotic heterogeneity without worsening outcomes [45, 46]. A recent study among pediatric patients with febrile neutropenia specifically demonstrated a significant decrease in extended-spectrum β -lactamase producers in blood and stool cultures after implementation of antibiotic cycling over a 3-year period [47].

 Table 39.5
 Typical empiric antibiotic therapy for patients with febrile neutropenia

Inpatient therapy	
(intravenous)	Outpatient therapy (low risk) (per os)
Cefepime	Ciprofloxacin ^a + amoxicillin/
	clavulanate
Imipenem/cilastatin	Moxifloxacin ^a
Meropenem	Levofloxacin ^a
Piperacillin/tazobactam	
Ceftazidime ^b	
Vancomycin ^c	

^aPatients previously on fluoroquinolone prophylactic therapy are not candidates for oral antibiotic therapy

^bWeak gram-positive coverage may limit utility

^cConsider vancomycin in patients who have a history of methicillinresistant *Staphylococcus aureus* (MRSA) colonization, clinical instability, soft tissue infection with high risk of MRSA, or indwelling intravenous catheters

If a patient is risk-stratified into a low-risk cohort after utilizing a validated scoring method, it is appropriate to consider treatment with oral antibiotics. First-line oral therapy typically consists of ciprofloxacin and amoxicillin/clavulanate. Alternative monotherapies with either moxifloxacin or levofloxacin are acceptable. Of note, if a patient was on prophylactic therapy with a fluoroquinolone prior to presentation, these oral regiments should not be utilized. Several studies have demonstrated non-inferiority of oral versus intravenous antibiotics in low-risk patients with febrile neutropenia [48-50]. Given increasing reports of adverse reactions to fluoroquinolone therapy (tendinopathies, CNS complications, etc.), it is important to discuss these potential risks with patients discharged on these oral agents [51, 52]. It is recommended that patients who are identified as potential candidates for home therapy are observed for a period of time in the ED prior to discharge (4-12 h).

Antifungal Therapy

Empiric antifungal therapy is not typically recommended in the acute setting, unless the patient has known invasive fungal disease or has physical exam findings suggestive of invasive fungal disease (i.e., mucormycosis). In general, antifungal therapy is initiated only after 4 or more days (average 7–10 days) of persistent fevers despite broadspectrum antibiotic therapy [53, 54]. Those with the highest risk of mold infections (neutropenia >10 days, allogenic stem cell recipient, prolonged steroid use) should receive early antifungal coverage (4 days of persistent fever), unless they are already on prophylactic antifungal medications [55].

Antiviral Therapy

Similar to antifungal therapy, antiviral therapy is not typically recommended in the acute setting, unless there is clear indication of a viral infection prompting treatment (i.e., HSV, VZV). It is prudent to assess for common viral upper respiratory infections with a respiratory viral polymerase chain reaction (PCR) swab. Additionally, assessing and treating for influenza may be indicated, especially if a patient presents with the appropriate constellation of symptoms [7].

Granulocyte Colony-Stimulating Factor (GCSF)

Although there is a role for the use of GCSF to prevent febrile neutropenia in patients receiving myelosuppressive therapy, the use of GCSF in patients with established febrile neutropenia has been controversial. Overall, the medication appears to have a negligible effect on patient-centered outcomes [56]. Specifically, GCSF given during febrile neutropenia does not improve overall patient mortality [57]. Conversely, patients receiving GCSF typically have a shorter duration of neutropenia, faster recovery from fever, and shorter duration of antibiotic use.

Future Considerations

Biomarkers

A significant proportion of patients with febrile neutropenia have little or no symptoms to assist clinicians in making a definitive diagnosis. Although some patients may have noninfectious causes of fever (tumor fever, drug fever) or minor viral infections, there is a subset of febrile neutropenic patients who appear clinically well, despite bacteremia. Unfortunately, blood cultures are classically insensitive and are prone to false positives due to contamination with skin flora. This has led researchers to explore several biomarkers as potential risk stratifications tools to predict poor outcomes in patients with febrile neutropenia.

Procalcitonin (PCT) has emerged as a valuable biomarker to help clinicians differentiate between bacterial illness and alternative causes of inflammation/infection. Unfortunately, previous literature has demonstrated a high variability in PCT levels among patients with febrile neutropenia, and there have been reports of bacteremia in patients with low procalcitonin levels [58, 59]. Serial PCT levels may provide improved discriminatory power [60].

CRP has also been proposed as a possible biomarker to aide in risk stratification. Overall, CRP may have a slightly higher sensitivity than procalcitonin (using a cutoff of >20 mg/L for CRP and >0.2 ng/ml for PCT). However, this sensitivity (0.82, 95% CI) is still too low to safely utilize CRP alone to risk stratify patients. Additionally, PCT appeared to be more discriminatory than CRP at time of disposition.

Interleukin-6 (IL-6) and interleukin-8 (IL-8) may also provide some insight into the risk of severe febrile neutropenia. Both IL-6 and IL-8 appear to have better predictive value than CRP for high-risk febrile neutropenia, with one study reporting that low IL-8 levels predicted patients at low risk for bacteremia with a sensitivity of 0.9 and a negative predictive value of 0.98 [61, 62].

Overall, biomarkers may prove to be very important in differentiating high- vs low-risk febrile neutropenia. Currently, however, the body of literature is lacking, and more evidence is needed before utilizing a single biomarker (or a combination of biomarkers) to definitely risk stratify patients.

Conclusion

Febrile neutropenia is a common complication of cytotoxic chemotherapy, and despite decades of study, it continues to present challenges for acute care providers. It is imperative to initiate a broad workup for these patients and provide rapid, broad-spectrum antibiotic therapy. For clinically stable patients, it is appropriate to apply validated risk stratification tools to identify a low-risk cohort appropriate for discharge. For patients qualifying for outpatient management, one must assure that the proper outpatient structure is in place prior to discharge. Treatment with antibiotics should cover both typical and atypical organisms and should account for previous cultures as well as the bacterial sensitivities of a particular hospital or region. As biomarkers for disease severity are developed and validated, we may more objectively identify patients at higher or lower risk for severe disease.

References

- Taplitz RA, Kennedy EB, Bow EJ, Crews J, Gleason C, Hawley DK, et al. Outpatient management of fever and neutropenia in adults treated for malignancy: American Society of Clinical Oncology and Infectious Diseases Society of America clinical practice guideline update. J Clin Oncol. 2018;36(14):1443–53.
- Carmona-Bayonas A, Jiménez-Fonseca P, Virizuela Echaburu J, Antonio M, Font C, Biosca M, Ramchandani A, et al. Prediction of serious complications in patients with seemingly stable febrile neutropenia: validation of the clinical index of stable febrile neutropenia in a prospective cohort of patients from the FINITE study. J Clin Oncol. 2015;33(5):465–71.
- Kuderer NM, Dale DC, Crawford J, Cosler LE, Lyman GH. Mortality, morbidity, and cost associated with febrile neutropenia in adult cancer patients. Cancer. 2006;106(10):2258–66.
- 4. Pizzo PA. Management of fever in patients with cancer and treatmentinduced neutropenia. N Engl J Med. 1993;328(18):1323–32.
- Zimmer AJ, Freifeld AG. Optimal management of neutropenic fever in patients with cancer. J Oncol Pract. 2019;15(1):19–24.
- Valent P. Low blood counts: immune mediated, idiopathic, or myelodysplasia. Hematology Am Soc Hematol Educ Program. 2012;2012:485–91.
- Freifeld AG, Bow EJ, Sepkowitz KA, Boeckh MJ, Ito JI, Mullen CA, Raad II, Rolston KV, Young JA, Wingard JR, Infectious Diseases Society of America. Clinical practice guideline for the use of antimicrobial agents in neutropenic patients with cancer: 2010 update by the Infectious Diseases Society of America. Clin Infect Dis. 2011;52(4):e56–93.
- Lewis MA, Hendrickson AW, Moynihan TJ. Oncologic emergencies: pathophysiology, presentation, diagnosis, and treatment. CA Cancer J Clin. 2011;61(5):287–314.
- Crawford J, Dale DC, Lyman GH. Chemotherapy-induced neutropenia: risks, consequences, and new directions for its management. Cancer. 2004;100(2):228–37.
- Nesher L, Rolston KV. The current spectrum of infection in cancer patients with chemotherapy related neutropenia. Infection. 2014;42(1):5–13.

- Lyman GH, Abella E, Pettengell R. Risk factors for febrile neutropenia among patients with cancer receiving chemotherapy: a systematic review. Crit Rev Oncol Hematol. 2014;90(3):190–9.
- Lyman GH, Dale DC, Friedberg J, Crawford J, Fisher RI. Incidence and predictors of low chemotherapy dose-intensity in aggressive non-Hodgkin's lymphoma: a nationwide study. J Clin Oncol. 2004;22(21):4302–11.
- Klastersky J, Ameye L, Maertens J, Georgala A, Muanza F, Aoun M, et al. Bacteraemia in febrile neutropenic cancer patients. Int J Antimicrob Agents. 2007;30(Suppl 1):S51–9.
- Zell JA, Chang JC. Neoplastic fever: a neglected paraneoplastic syndrome. Support Care Cancer. 2005;13(11):870–7.
- Bodey GP, Jadeja L, Elting L. Pseudomonas bacteremia. Retrospective analysis of 410 episodes. Arch Intern Med. 1985;145(9):1621–9.
- Wisplinghoff H, Seifert H, Wenzel RP, Edmond MB. Current trends in the epidemiology of nosocomial bloodstream infections in patients with hematological malignancies and solid neoplasms in hospitals in the United States. Clin Infect Dis. 2003;36(9):1103–10.
- Holland T, Fowler VG Jr, Shelburne SA 3rd. Invasive gram-positive bacterial infection in cancer patients. Clin Infect Dis. 2014;59(Suppl 5):S331–4.
- Gudiol C, Bodro M, Simonetti A, Tubau F, González-Barca E, Cisnal M, et al. Changing aetiology, clinical features, antimicrobial resistance, and outcomes of bloodstream infection in neutropenic cancer patients. Clin Microbiol Infect. 2013;19(5):474–9.
- Portugal RD, Garnica M, Nucci M. Index to predict invasive mold infection in high-risk neutropenic patients based on the area over the neutrophil curve. J Clin Oncol. 2009;27(23):3849–54.
- Pagano L, Caira M, Candoni A, Offidani M, Fianchi L, Martino B, et al. The epidemiology of fungal infections in patients with hematologic malignancies: the SEIFEM-2004 study. Haematologica. 2006;91(8):1068–75.
- 21. von Lilienfeld-Toal M, Berger A, Christopeit M, Hentrich M, Heussel CP, Kalkreuth J, et al. Community acquired respiratory virus infections in cancer patients-guideline on diagnosis and management by the Infectious Diseases Working Party of the German Society for Haematology and Medical Oncology. Eur J Cancer. 2016;67:200–12.
- Hirsch HH, Martino R, Ward KN, Boeckh M, Einsele H, Ljungman P. Fourth European Conference on Infections in Leukaemia (ECIL-4): guidelines for diagnosis and treatment of human respiratory syncytial virus, parainfluenza virus, metapneumovirus, rhinovirus, and coronavirus. Clin Infect Dis. 2013;56(2):258–66.
- Saral R, Ambinder RF, Burns WH, Angelopulos CM, Griffin DE, Burke PJ, Lietman PS. Acyclovir prophylaxis against herpes simplex virus infection in patients with leukemia. A randomized, double-blind, placebo-controlled study. Ann Intern Med. 1983;99(6):773–6.
- McKenzie H, Hayes L, White K, Cox K, Fethney J, Boughton M, Dunn J. Chemotherapy outpatients' unplanned presentations to hospital: a retrospective study. Support Care Cancer. 2011;19(7):963–9.
- Ciuraru NB, Braunstein R, Sulkes A, Stemmer SM. The influence of mucositis on oral thermometry: when fever may not reflect infection. Clin Infect Dis. 2008;46(12):1859–63.
- Heussel CP, Kauczor HU, Heussel GE, Fischer B, Begrich M, Mildenberger P, Thelen M. Pneumonia in febrile neutropenic patients and in bone marrow and blood stem-cell transplant recipients: use of high-resolution computed tomography. J Clin Oncol. 1999;17(3):796–805.
- Talcott JA, Siegel RD, Finberg R, Goldman L. Risk assessment in cancer patients with fever and neutropenia: a prospective, two-center validation of a prediction rule. J Clin Oncol. 1992;10(2):316–22.
- Bodey GP, Buckley M, Sathe YS, Freireich EJ. Quantitative relationships between circulating leukocytes and infection in patients with acute leukemia. Ann Intern Med. 1966;64(2):328–40.

- 29. Talcott JA, Finberg R, Mayer RJ, Goldman L. The medical course of cancer patients with fever and neutropenia. Clinical identification of a low-risk subgroup at presentation. Arch Intern Med. 1988;148(12):2561–8.
- 30. Klastersky J, Paesmans M, Rubenstein EB, Boyer M, Elting L, Feld R, et al. The Multinational Association for Supportive Care in Cancer risk index: a multinational scoring system for identifying low-risk febrile neutropenic cancer patients. J Clin Oncol. 2000 Aug;18(16):3038–51.
- Baden LR, Swaminathan S, Angarone M, Blouin G, Camins BC, Casper C, et al. Prevention and treatment of cancer-related infections, version 2.2016, NCCN clinical practice guidelines in oncology. J Natl Compr Cancer Netw. 2016;14(7):882–913.
- 32. Moon H, Choi YJ, Sim SH. Validation of the Clinical Index of Stable Febrile Neutropenia (CISNE) model in febrile neutropenia patients visiting the emergency department. Can it guide emergency physicians to a reasonable decision on outpatient vs. inpatient treatment? PLoS One. 2018;13(12):e0210019.
- Mohindra R, Mathew R, Yadav S, Aggarwal P. CISNE versus MASCC: Identifying low risk febrile neutropenic patients. Am J Emerg Med. 2020;38(11):2259–63.
- 34. Zheng B, Toarta C, Cheng W, Taljaard M, Reaume N, Perry JJ. Accuracy of the Multinational Association of Supportive Care in Cancer (MASCC) and Clinical Index of Stable Febrile Neutropenia (CISNE) scores for predicting serious complications in adult patients with febrile neutropenia: a systematic review and meta-analysis. Crit Rev Oncol Hematol. 2020;149:102922.
- 35. Haeusler GM, Thursky KA, Slavin MA, Mechinaud F, Babl FE, Bryant P, et al. External validation of six pediatric fever and neutropenia clinical decision rules. Pediatr Infect Dis J. 2018;37(4):329–35.
- 36. Haeusler GM, Thursky KA, Slavin MA, Babl FE, De Abreu Lourenco R, Allaway Z, Australian PICNICC Study Group and the PREDICT Network. Risk stratification in children with cancer and febrile neutropenia: a national, prospective, multicentre validation of nine clinical decision rules. EClinicalMedicine. 2020;100220:18.
- Castagnola E, Dufour C. Clinical decision rules for infectious risk stratification of children with febrile neutropenia: are we looking for the Yeti? EClinicalMedicine. 2020;100262:19.
- Phillips RS, Lehrnbecher T, Alexander S, Sung L. Updated systematic review and meta-analysis of the performance of risk prediction rules in children and young people with febrile neutropenia. PLoS One. 2012;7(5):e38300.
- Rosa RG, Goldani LZ. Cohort study of the impact of time to antibiotic administration on mortality in patients with febrile neutropenia. Antimicrob Agents Chemother. 2014;58(7):3799–803.
- Perron T, Emara M, Ahmed S. Time to antibiotics and outcomes in cancer patients with febrile neutropenia. BMC Health Serv Res. 2014;14:162.
- 41. Daniels LM, Durani U, Barreto JN, O'Horo JC, Siddiqui MA, Park JG, Tosh PK. Impact of time to antibiotic on hospital stay, intensive care unit admission, and mortality in febrile neutropenia. Support Care Cancer. 2019;27(11):4171–7.
- 42. Rosa RG, Goldani LZ, dos Santos RP. Association between adherence to an antimicrobial stewardship program and mortality among hospitalised cancer patients with febrile neutropaenia: a prospective cohort study. BMC Infect Dis. 2014;14:286.
- 43. O'Horo JC, Marcelin JR, Abu Saleh OM, Barwise AK, Odean PM, Rivera CG, et al. Standardizing febrile neutropenia management: antimicrobial stewardship in the hematologic malignancy population. J Oncol Pract. 2019;15(9):e843–8.
- 44. Martinez-Nadal G, Puerta-Alcalde P, Gudiol C, Cardozo C, Albasanz-Puig A, Marco F, et al. Inappropriate empirical antibiotic treatment in high-risk neutropenic patients with bacteremia in the era of multidrug resistance. Clin Infect Dis. 2020;70(6):1068–74.

- 45. Tan BH, Guzman MRT, Donato LKS, Kalimuddin S, Lee WHL, Tan AL, Wong GC. Impact of an alternating first-line antibiotics strategy in febrile neutropenia. PLoS One. 2018;13(11):e0208039.
- 46. Chong Y, Shimoda S, Yakushiji H, Ito Y, Miyamoto T, Kamimura T, et al. Antibiotic rotation for febrile neutropenic patients with hematological malignancies: clinical significance of antibiotic heterogeneity. PLoS One. 2013;8(1):e54190.
- 47. Teranishi H, Koga Y, Nishio H, Kato W, Ono H, Kanno S, et al. Clinical efficacy of cycling empirical antibiotic therapy for febrile neutropenia in pediatric cancer patients. J Infect Chemother. 2017;23(7):463–7.
- Vidal L, Paul M, Ben-Dor I, Pokroy E, Soares-Weiser K, Leibovici L. Oral versus intravenous antibiotic treatment for febrile neutropenia in cancer patients. Cochrane Database Syst Rev. 2004;(4):CD003992. Update in: Cochrane Database Syst Rev. 2013;10:CD003992.
- 49. Petrilli AS, Dantas LS, Campos MC, Tanaka C, Ginani VC, Seber A. Oral ciprofloxacin vs. intravenous ceftriaxone administered in an outpatient setting for fever and neutropenia in low-risk pediatric oncology patients: randomized prospective trial. Med Pediatr Oncol. 2000;34(2):87–91.
- Niho S, Ohe Y, Goto K, Ohmatsu H, Matsumoto T, Kubota K, et al. Randomized trial of oral versus intravenous antibiotics in low-risk febrile neutropenic patients with lung cancer. Jpn J Clin Oncol. 2004;34(2):69–73.
- Alves C, Mendes D, Marques FB. Fluoroquinolones and the risk of tendon injury: a systematic review and meta-analysis. Eur J Clin Pharmacol. 2019;75(10):1431–43.
- Odeh M, Kogan Y, Paz A, Elias N. Delirium induced by levofloxacin. J Clin Neurosci. 2019;66:262–4.
- Pizzo PA, Robichaud KJ, Gill FA, Witebsky FG. Empiric antibiotic and antifungal therapy for cancer patients with prolonged fever and granulocytopenia. Am J Med. 1982;72(1):101–11.

- Wingard JR. Empirical antifungal therapy in treating febrile neutropenic patients. Clin Infect Dis. 2004;39(Suppl 1):S38–43.
- 55. Staber P, Langner S, Dornbusch HJ, Neumeister P. Antifungal management in cancer patients. Wien Med Wochenschr. 2007;157(19–20):503–10.
- Garcia-Carbonero R, Mayordomo JI, Tornamira MV, et al. Granulocyte colony-stimulating factor in the treatment of high-risk febrile neutropenia: a multicenter randomized trial. J Natl Cancer Inst. 2001;93(1):31–8.
- 57. Mhaskar R, Clark OA, Lyman G, Engel Ayer Botrel T, Morganti Paladini L, Djulbegovic B. Colony-stimulating factors for chemotherapy-induced febrile neutropenia. Cochrane Database Syst Rev. 2014;(10):CD003039.
- Giamarellou H, Giamarellos-Bourboulis EJ, Repoussis P, et al. Potential use of procalcitonin as a diagnostic criterion in febrile neutropenia: experience from a multicentre study. Clin Microbiol Infect. 2004;10(7):628–33.
- 59. Robinson JO, Lamoth F, Bally F, Knaup M, Calandra T, Marchetti O. Monitoring procalcitonin in febrile neutropenia: what is its utility for initial diagnosis of infection and reassessment in persistent fever? PLoS One. 2011;6(4):e18886.
- Reitman AJ, Pisk RM, Gates JV 3rd, Ozeran JD. Serial procalcitonin levels to detect bacteremia in febrile neutropenia. Clin Pediatr (Phila). 2012;51(12):1175–83.
- Cost CR, Stegner MM, Leonard D, Leavey P. IL-8 predicts pediatric oncology patients with febrile neutropenia at low risk for bacteremia. J Pediatr Hematol Oncol. 2013;35(3):206–11.
- 62. Chaudhary N, Kosaraju K, Bhat K, Bairy I, Borker A. Significance of interleukin-6 (IL-6) and C-reactive protein (CRP) in children and young adults with febrile neutropenia during chemotherapy for cancer: a prospective study. J Pediatr Hematol Oncol. 2012;34(8):617–23.



Neutropenic Enterocolitis

Steven McGaughey and Robert L. Cloutier

Case Study

A 70-year-old man with acute myeloid leukemia on daunorubicin and cytarabine for induction therapy presents to the emergency department (ED) after suddenly developing fever, diffuse abdominal pain, and diarrhea. On review of his chart, it is noted that he has been neutropenic (absolute neutrophil count (ANC) = 350/dL since 9 days prior to presentation. Today, vitals are notable for a respiratory rate of 19 breaths per minute, sinus tachycardia of 120 beats/min, and blood pressure of 105/60 mmHg. On exam, he has right lower quadrant abdominal pain with associated involuntary guarding. Laboratory evaluation reveals a white blood cell count of 1200 cells/ μ L (ANC = 100/dL), a hematocrit of 23%, and 40,000 platelets/µL. Computed tomography (CT) scan of the abdomen and pelvis shows thickening of the colonic wall with diffuse pericolic edema and pneumatosis. Since the differential diagnosis includes a strong suggestion of neutropenic enterocolitis, the hematology and surgical teams are promptly consulted. The patient is admitted and started on intravenous fluids, ceftazidime and metronidazole, as well as granulocyte colony-stimulating factor (G-CSF). The next day, the patient becomes hemodynamically unstable and is transferred to the intensive care unit where he remains on broad-spectrum antibiotics with close monitoring. On hospital day 5, he shows clinical improvement. Blood cultures remain no growth, and stool studies are negative for Clostridium difficile (C. difficile) or other enteropathogenic organisms. With continued conservative management, the patient had full resolution of symptoms following 14 days of treatment.

Department of Emergency Medicine, Oregon Health & Science University, Portland, OR, USA e-mail: mcgaughe@ohsu.edu

R. L. Cloutier

Introduction

Neutropenic enterocolitis (NE) is a poorly understood and life-threatening complication of severe neutropenia [1]. While it is most often seen in patients with hematologic malignancies, it can be seen in a wide range of disease processes. NE was first reported as a complication of treatment of childhood leukemias and remains most common in pediatric patients [2]. However, NE is increasingly documented in adults as well, especially in those patients on intensive chemotherapies for hematologic malignancies. Incidence is variable with rates ranging from 0.8% to 26% with a pooled incidence rate of 5.3% [3-5]. Clinical manifestations of NE include fever, abdominal pain or distension, diarrhea, and intestinal bleeding. There are no pathognomonic manifestations or strict diagnostic criteria; thus, a high level of clinical acumen is required to make a prompt diagnosis and start appropriate treatment [1].

NE most commonly occurs 2–3 weeks after chemotherapy is initiated. At this time, a combination of impaired mucosal defense and intestinal injury leaves the bowel susceptible to bacterial invasion with possible progression to perforation [1, 6, 7]. If untreated, these patients can quickly progress to sepsis syndrome with resultant multisystem organ failure. Early reports of NE in the 1960s were almost uniformly fatal. Over the subsequent decades, outcomes have improved; however, morbidity and mortality remain high at between 50% and 100% [8–12].

With the advent of new chemotherapies, more patients than ever are at risk for developing NE, including those with both hematologic malignancies and solid tumors [1, 13]. Emergency physicians and other frontline medical professionals are critical in rapidly identifying and treating neutropenic patients. Given the high morbidity of NE, it is essential that these clinicians have familiarity with this disease process to allow them to more promptly detect and treat these patients appropriately.

This chapter will discuss the background, clinical presentation, diagnostic criteria, risk factors, pathophysiology,

S. McGaughey (⊠)

Pediatric Emergency Medicine, Department of Emergency Medicine, Office of Admissions, School of Medicine, Oregon Health & Science University, Portland, OR, USA

workup, and management of NE. We note that the nomenclature of NE is heterogeneous; typhlitis, ileocecitis, and cecitis have all been used synonymously in the literature to describe NE. For clarity, we have chosen to use the term NE exclusively in this chapter. This should not be confused with necrotizing enterocolitis or NEC, a separate and unrelated neonatal disease.

Background and State of the Evidence

The first description of NE dates back at to 1933, when Cooke noted submucosal hemorrhage and appendiceal perforation in children with leukemia [2]. More contemporary reports date to 1962, with a case series by Amromin and Salomon [14]. As noted in a review by Cunningham, these patients almost uniformly experienced abdominal pain, distension, leukopenia, and intestinal necrosis with histologic evidence of bacterial invasion of the mucosal and submucosal bowel wall [8]. These patients characteristically developed necrotizing enteric lesions shortly after intensive chemotherapy. These lesions developed regardless of whether the patient responded to chemotherapy. In fact, most patients with NE actually experience excellent therapeutic response prior to developing NE. Unfortunately, the subsequent clinical course in patients diagnosed with NE in the 1960s was marked by hemodynamic collapse and death.

Since then, there have been numerous reports of NE across a spectrum of patients, both with and without cancer. However, the current literature suffers from many limitations. First, these articles have mainly been restricted to case reports and series. A recent review by Gorschlüter noted that, of the 145 papers on neutropenic enterocolitis, there were no clinical trials or case-control studies. The few remaining prospective studies have significant limitations and meet only level 3b evidence according to the Oxford Center for Evidence-Based Medicine [5]. Second, there are no unified diagnostic criteria for NE. As a result, the overall grade of the evidence in neutropenic enterocolitis is low. An editorial by Gorbach summed up the challenges of understanding NE: "the diversity of the pathology is matched by the difficulty in establishing the diagnosis on the basis of only clinical findings" [9]. Despite these challenges, clinicians have made progress in early recognition and treatment of NE, leading to meaningful improvements in patient outcomes [1].

Clinical Presentation

Most patients with NE will present with a combination of diffuse crampy abdominal pain, vomiting, abdominal distension, and diarrhea [1, 10, 15, 16]. The location of abdominal pain is most often diffuse with or without rebound tender-

 Table
 40.1
 Common signs and symptoms of neutropenic enterocolitis

Vague diffuse crampy abdominal pain
With or without peritoneal signs including rebound and/or guarding
If localizes: more commonly to the right lower quadrant
Abdominal distension
Diarrhea
Fever
Melanotic stools or hematochezia

ness. Some patients do have localized abdominal pain when NE affects a specific segment of bowel. For example, patients frequently experience right lower quadrant pain which may be due to distension of, and limited blood supply to, the cecum [10, 15]. Patients can have hematochezia or melanotic stools; however, the frequency of these presentations is varied [7, 11, 17, 18]. Peritoneal signs, such as abdominal rebound, guarding, or abdominal distension, may be present if patients develop intestinal perforation. While fever is common, it may be absent in severely immunocompromised patients [10, 15, 16, 19, 20].

Overall, the clinical presentation of NE is variable with significant overlap between NE and many other abdominal pathologies representing infectious, chemotherapeutic, and malignancy-induced as well as primary and surgical conditions or causes. The differential diagnosis includes infections, such as *C. difficile* colitis, and chemotherapy-induced conditions, such as mucositis, mycophenolate injury, as well as graft-versus-host disease (GVHD); malignancy-induced conditions, such as recurrent lymphoma and leukemic infiltration; and, lastly, surgical conditions, such as appendicitis or intussusception [1, 7, 12, 21]. Clinically, it is difficult to differentiate between these conditions; thus, prompt imaging is crucial to diagnose NE.

Common presenting signs and symptoms are summarized in Table 40.1.

Diagnostic Criteria

There are currently no universal diagnostic criteria for NE. This is largely due to the vast heterogeneity in pathological, radiographic, and clinical diagnostic criteria across the literature. In 2005, Gorschlüter published a systematic review of NE literature and proposed the diagnostic criteria listed in Table 40.2 [5], which include clinical signs, such as fever and abdominal pain, in combination with imaging findings. While these criteria have been generally accepted, some suggest refining the criteria to include the presence of neutropenia (ANC < 500 × 106 cells/L) and to rule out alternative conditions, such as GVHD, *C. difficile* colitis, and appendicitis [1]. Still others debate including fever as a

 Table
 40.2
 Suggested
 diagnostic
 criteria
 for
 neutropenic

 enterocolitis

 neutropenic

 </td

Presence of fever (axillary temperature >38.0 °C or rectal temperature >38.5 °C)

Abdominal pain (at least degree 3 determined by the patient using a visual analogous scale pain score ranging from degree 1 to 10) Demonstration of the bowel wall thickening of more than 4 mm (transversal scan) over more than 30 mm (longitudinal scan) in any segment by US or CT

From Gorschlüter et al. [5], with permission John Wiley US ultrasound, CT computed tomography

criteria given how poorly fever correlates with the ultimate diagnosis of NE [22]. Regardless, most authors broadly agree that a combination of clinical signs and symptoms with bowel wall thickening (BWT) on imaging is most appropriate to ultimately diagnose NE.

NE remains a difficult diagnosis for frontline clinicians, including emergency physicians. Neutropenic patients on chemotherapy are at high risk to develop many possible complications. Consequently, clinicians must perform a thoughtful and thorough evaluation and have a high index of suspicion for NE in neutropenic patients presenting with abdominal pain.

Risk Factors

There are several risk factors for developing NE. Numerous chemotherapeutic agents are strongly associated with NE. Cyclophosphamide, 5-fluorouracil, and its prodrug capecitabine, ifosfamide, cisplatin, carboplatin, gemcitabine, cytosine arabinoside, vincristine, doxorubicin, idarubicin, leucovorin, and daunorubicin are all associated with NE [10, 13, 15, 23–25]. Additionally, taxanes (docetaxel, paclitaxel) and vinorelbine, which are used for solid tumors, have been implicated in recent reports [1].

Beyond the chemotherapeutic agents themselves, the risk of a patient developing NE changes depending on the timing of chemotherapy and the resultant duration of neutropenia. The average time between chemotherapy and development of NE ranges from 12 to 17 days. It is during this time frame when cytotoxic chemotherapy causes maximal injury of the mucosal barrier and makes NE possible [26–28]. Some patients will develop NE later in their chemotherapeutic course; however, most NE patients will have neutropenia 2–9 days before developing symptoms [7, 11, 15]. While later onset is less common, this specific population is at an increased risk of death [11]. Other factors associated with increased mortality include older age, more severe and prolonged neutropenia, and concomitant systemic infections [11].

Lastly, there is an association between the type of malignancy and NE. Early cases of NE were found almost exclusively in pediatric leukemias. Today, while the majority of cases of NE are still found in patients with leukemia and lymphoma, a growing number of NE cases are found in patients with solid tumors including breast, lung, colorectal, and ovarian cancer [10, 13, 23, 24]. Overall, patients with hematologic malignancies who develop NE portend a worse overall survival. This may be linked to the longer duration and more profound neutropenia seen with these malignancies and their associated treatments [11].

While most NE is associated with specific chemotherapies, there are a small number of reported cases in patients prior to initiation of treatment [17, 29, 30]. Additionally, there are rare cases of non-oncologic conditions leading to NE. A single trauma patient developed NE in the setting of reversible neutropenia secondary to nafcillin therapy for osteomyelitis [19]. Other cases have been reported in the setting of aplastic anemia, cyclic neutropenia, acquired immunodeficiency syndrome, and immune-suppression for bone marrow or renal transplantation [19, 20]. Overall, any condition which causes prolonged neutropenia places patients at risk for developing NE.

Pathophysiology

The pathophysiology of NE is multifactorial and incompletely understood. The primary effects of neutropenia decrease the immunologic response against intestinal invasion, making the patient more susceptible to infection. Secondly, the intestinal wall is damaged through a number of mechanisms: direct destruction due to chemotherapy and/or radiation, leukemic or lymphomatous infiltrates, or intramural hemorrhage secondary to severe thrombocytopenia. This compromised mucosa ultimately allows translocation of normal gut microbial flora [1, 6, 7, 11, 31]. Innate immunity of the submucosal tissue and proinflammatory mediators leads to a cascade of events culminating in epithelial cell apoptosis and mucosal permeability. Visualized bowel is thickened and edematous with hemorrhage and ulceration [1]. As noted previously, the cecum is particularly susceptible to injury secondary to its more limited blood supply. However, while NE most commonly affects the cecum, this condition may affect any portion of the intestinal tract.

The combination of mucosal injury and neutropenia places patients at high risk of developing bacteremia from gut flora, including *Enterococcus*, *Klebsiella*, and coagulase-negative *Staphylococcus* [11, 12]. In addition, *C. difficile* is commonly seen in NE, but it is generally not thought to be pathogenic as there is no mortality difference in patients with or without this pathogen [32]. Invasive fungal infections, mostly commonly by *Candida*, occur in approximately 5% of cases [33]. Bacteremia is more common, with positive blood cultures found in 28–80% of definite cases of

NE [7, 15, 34]. Overall, while the most common single isolate from blood cultures are aerobic gram-negative bacilli, most cultures show polymicrobial infections [7, 11, 35]. There is no data to suggest that a specific organism or combination of pathogens is associated with NE.

Pediatric Considerations

Considerations in the pediatric population closely mirror those of adults. Hematopoietic malignancies such as acute lymphoid leukemia (ALL) and acute myeloid leukemia (AML) are most commonly associated with NE in the pediatric population [36, 37]. The incidence has been reported to be higher in pediatric patients, varying between 2.6% and 16.2%, which may reflect the highly aggressive malignancies and associated intensive chemotherapy regimens used for treatment [36–39]. A study by McCarville characterizing NE in pediatric patients noted that, similar to adults, diagnosing NE in children based on symptoms alone, is challenging [37]. The clinical presentation of NE in pediatric patients is similar to adults and includes abdominal tenderness, fever, diarrhea, and nausea. However, in 9-16% of cases, typical signs and symptoms, including abdominal pain, fever, and neutropenia, were absent [37]. Overall age, race, and gender are poorly associated with risk of NE; however, patients older than 16 years had an increased odds ratio of NE and had poorer response to therapy than younger patients. Medications and chemotherapeutic agents associated with pediatric NE include GCS-F, topotecan, atovaquone, PEG-L-asparaginase, idarubicin, cytosine arabinoside, hydrocortisone, methotrexate, and carboplatin. Additionally, treatment with cytarabine and presenting with abdominal distention are poor prognostic factors, which are associated with increased mortality [36].

Workup

Initial workup for NE requires ruling out other common conditions, including *C. difficile* colitis, GVHD, and other abdominal pathology, while meaningfully addressing the broader threats posed by potential neutropenia. Thus, initial laboratory testing, including complete blood count, coagulation studies, chemistry panel, and *C. difficile* toxin or polymerase chain reaction (PCR) assay, is recommended. While no laboratory tests will confirm the diagnosis of NE, presence of neutropenia in the absence of other abdominal pathologies strongly suggests a patient is at risk for NE. In these patients, additional imaging is critical in making the correct diagnosis and starting appropriate treatment.

The Role of Diagnostic Imaging: CT and Ultrasound

Because the clinical presentation is nonspecific, imaging is essential in the diagnosis of NE. Computed tomography (CT) and ultrasonography (US) are the most common modalities for diagnosing NE. Conversely, plain radiographs are of limited utility; while positive findings, such as paucity of air in the cecum with associated pneumatosis, can be helpful, most x-rays are nonspecific or normal.

Many clinicians favor CT to evaluate for NE. CT provides a more detailed view of bowel integrity and better characterizes surrounding abdominal inflammation and bowel wall thickening (BWT) (Fig. 40.1) [12, 16, 27, 40, 41]. In addition, CT provides better visualization of other radiographic features of abdominal pathology, including pneumatosis and nodularity. This becomes clinically important when trying to differentiate between NE and other mimics. A retrospective



Fig. 40.1 NE imaging findings. (**a**, **b**) Computed tomography images from two different patients with NE; both show cecal wall thickening and dilation (*brackets*). (**c**) Colonic dilatation on an abdominal radio-

graph; findings are most prominent in the cecum (*bracket*). (From Sachak et al. [12], with permission Wolters Kluwer)

study characterized the CT features of neutropenic patients with radiographic bowel abnormalities—noting specific patterns of BWT, pneumatosis, and nodularity on CT—useful in distinguishing NE from other pathologies, including GVHD, cytomegalovirus colitis, and *C. difficile* colitis [41]. NE was also best characterized by its predilection for the right colon and cecum, with occasional involvement of the small bowel. Lastly, NE exhibited the greatest degree of pneumatosis and mesenteric stranding when compared with GVHD or *C. difficile* colitis [41].

While CT provides highly detailed images, ultrasound can be deployed more quickly and avoids radiation. In an emergent setting, ultrasound-derived BWT measurements can be used to help support the diagnosis of NE or rule out other diagnoses. Measurement criteria for BWT vary due to differences in the sensitivity of sonographic machines and the small number of published studies. Nonetheless, \geq 3–5 mm (outer wall to luminal surface) has often been defined as abnormal in pediatric and adult patients [10, 27, 36-38, 40, 42, 43]. Though most authors would agree that BWT of greater than 5 mm is abnormal, the prognostic significance of BWT remains unclear. A few studies, however, report that measurements of greater than 10 mm portend a poor outcome [15, 16, 27]. US is a particularly useful and radiation-free option in the pediatric population. In addition to more well-known signs, a unique finding in the pediatric population, pneumatosis intestinalis with portal venous gas, can be suggestive of pan-intestinal involvement [25].

US may also be useful as an efficient and inexpensive screening tool to initiate antibiotic therapy in the appropriate patient [44]. In a recent study, the authors used US to screen all hospitalized neutropenic patients with fever and abdominal pain and/or diarrhea to evaluate for signs of BWT. If present, they promptly started patients on antifungal agents, antibiotics, and granulocyte colony-stimulating factor. Their mortality of post-chemotherapy patients drastically reduced from 45% pre-intervention to 0% afterward, and mean diagnosis time after fever decreased from 9 days to 3 days [45].

For the emergency clinician, the most appropriate imaging modality will depend upon the hemodynamic stability of the patient. If a patient is safe to transport, he or she will benefit from a CT as this will provide more diagnostic information and help to differentiate from other serious complications of chemotherapy. However, in patients who are hemodynamically unstable, US is a powerful screening tool to evaluate for BWT at the bedside. For any patient who is critically ill, the emergency clinician should request emergent surgical consultation for possible operative resection of necrotic bowel.

Treatment

Medical Versus Surgical

There is no clear consensus on the most appropriate management of NE. Most patients will require supportive measures, including bowel rest, parenteral nutrition, and intravenous (IV) fluid support. Clinicians should also evaluate for and correct severe thrombocytopenia and coagulopathies [1]. In addition, we recommend the prudent advice by Gorschlüter: "to follow a pragmatic strategy analogous to neutropenic [fever of unknown origin], which is always treated as an infectious disease even if sometimes non-infectious causes are present" [5]. Unfortunately, there are no unified recommendations on the most appropriate antibiotics for NE patients [42, 46–49]. We recommend selecting antibiotics that cover a broad spectrum of gram-positive and gramnegative organisms as well as Pseudomonas and C. difficile. Monotherapies, such as piperacillin-tazobactam or imipenem-cilastatin, are reasonable initial choices. Duo therapies, combining another beta-lactam antipseudomonal agent with an aminoglycoside or, alternatively, cefepime or ceftazidime coupled with metronidazole, are also acceptable. Anaerobic coverage is important primarily when using a cephalosporin monotherapy [5]. Recommended doses for adults and children are summarized in Table 40.3 [51]. In

 Table 40.3
 Antibiotics for empiric treatment of adult and pediatric neutropenic enterocolitis

Antibiotics for empiric				
treatment	Dosages			
In adult neutropenic enterocolitis				
Monotherapy				
Piperacillin-tazobactam	3.375 g IV Q6 h			
Imipenem-cilastatin	500 mg IV Q6 h or 1 g IV Q6–8 h			
Duo therapy				
Ceftazidime	1 g IV Q8–12 h <i>or</i>			
Cefepime	2 g IV Q8 h <i>plus</i>			
Metronidazole	1 g IV Q6 h			
In pediatric (1–12 y of age) neutropenic enterocolitis				
Monotherapy				
Piperacillin-tazobactam	(>9 mo and <40 kg) 300 mg/kg/d IV divided Q8 h			
Imipenem-cilastatin	(>3 mo) 60–100 mg/kg/d IV divided Q6			
	h, max 2–4 g/d			
Duo therapy				
Ceftazidime	90–150 mg/kg/d IV divided Q8 h max			
	6 g/d <i>or</i>			
Cefepime	50 mg/kg IV Q8 h, max 2 g/dose <i>plus</i>			
Metronidazole	30 mg/kg/d IV divided Q6 h max 4 g/d			

From Cloutier [50], with permission Elsevier *IV* intravenous, *Q* every

Fig. 40.2 Suggested algorithm for the evaluation and treatment of patients with suspected neutropenic enterocolitis. See Table 40.1 and text for details. ANC absolute neutrophil count, CT computerized tomography, G-CSF granulocyte colonystimulating factor, GVHD graft-versus-host disease, PCR polymerase chain reaction, tpn total parenteral nutrition, US ultrasound. (Adapted from Nesher and Rolston, with permission Oxford University Press and the Infectious Diseases Society of America [1]). ^a see Table 40.2 for diagnostic criteria



Patient with suspected neutropenic enterocolitis

their invited review, Nesher discourages empiric use of antifungal medications as the frequency of invasive fungemia is only 5%. They further summarized their workup and treatment algorithm which is outlined in Fig. 40.2 [1, 33].

Surgical Intervention

Historically, autopsies of patients with NE showed high rates of transmural bowel necrosis and perforation [15]. Based on this, surgical intervention with bowel resection became firstline therapy, and early studies showed this reduced overall mortality. However, the clinical presentation of NE is heterogeneous, and experts today disagree on the necessity of early surgical intervention [5, 16, 51]. More recently, as clinicians
 Table 40.4
 Criteria for surgical intervention

Persistent gastrointestinal bleeding despite correction of
coagulopathies, thrombocytopenia, and neutropenia
Intraperitoneal free air
Deterioration despite sufficient medical management, suggesting
sepsis
Development of other surgical indications

gain more experience with NE, a greater proportion of patients have been successfully managed medically with broad-spectrum antibiotic therapy [5, 15, 16]. Now, surgery is typically reserved for more severe cases, as summarized in Table 40.4; however, there is still wide variation between institutions [1, 11, 12, 49, 52]. If a patient does require non-emergent surgical management, a recent review by Badgwell

suggests delaying surgery until neutropenia resolves as this lowers the overall mortality rate [22]. Overall, these recommendations are based on small case series studies; larger prospective studies are required to provide more definitive recommendations on medical versus surgical management.

If the emergency clinician suspects NE in a patient, he or she should organize critical care resources and consult surgery services early. Patients with NE are medically tenuous and have high potential to decompensate and require operative intervention.

Additional Therapy

Recombinant granulocyte colony-stimulating factor (G-CSF) has recently gained more widespread use to hasten bowel healing. In theory, G-CSF causes more rapid recovery of neutropenia which would therefore improve clinical outcomes [5, 10]. However, the data on G-CSF remains mixed. A recent large cohort study by Abu-Sbeih found that while G-CSF shortens the median time of neutropenia, it does not alter frequency of NE-related complications or mortality rates [11]. The American Society of Clinical Oncologic Clinical Practice Guidelines only recommend using a combination of antibiotics and empiric G-CSF in febrile neutropenic patients with high-risk factors including expected prolonged (>10 days) and profound (<0.1 × 109/L) neutropenia, age > 65 years, sepsis syndrome, or hospitalization at the time of fever development [53].

Despite recent gains in outcomes, morbidity and mortality remain high for NE. Treatment for a patient with NE is the same as that for other patients at risk for acute sepsis and hemodynamic collapse. An appropriate initial assessment should include venous access to support aggressive fluid support in addition to broad-spectrum antibiotics. Central venous access and an arterial line may be necessary for goal-directed therapy for sepsis. Lastly, clinicians should remain vigilant and monitor carefully for surgical indications, such as signs of an acute abdomen or hemodynamic compromise.

Summary

NE is a relatively rare but potentially devastating disease most often seen in neutropenic patients after chemotherapy. It is critical that emergency clinicians promptly identify and triage at-risk patients. Early recognition and intervention are essential to minimize morbidity and mortality in patients with NE. NE should be suspected in any neutropenic patient with abdominal pain. Patients who recently received chemotherapy are at particularly high risk. We recommend laboratory testing, including complete blood count to evaluate for neutropenia and thrombocytopenia, coagulation studies to evaluate for reversible coagulopathies, chemistry panel, and *C. difficile* toxin or PCR assay. In addition, we recommend rapid abdominal imaging. In a stable patient, CT can provide highly detailed information on bowel wall thickening which can help to differentiate between NE and other mimics. However, in the unstable patient, ultrasound may be appropriate as a fast, effective means to evaluate the bowel wall. Treatment includes bowel rest, parenteral nutrition, IV fluids, and correction of severe thrombocytopenia and coagulopathies. Broad-spectrum antibiotics with adequate coverage for intra-abdominal pathogens should be utilized. G-CSF may be beneficial in high-risk patients. Surgery services should be consulted early as NE patients can deteriorate rapidly.

References

- Nesher L, Rolston KVI. Neutropenic enterocolitis, a growing concern in the era of widespread use of aggressive chemotherapy. Clin Infect Dis. 2013;56(5):711–7. https://doi.org/10.1093/cid/cis998.
- 2. Cooke J. Acute leukemia in children. JAMA. 1933;101(6):432-5.
- Salazar R, Solá C, Maroto P, Tabernero JM, Brunet J, Verger G, et al. Infectious complications in 126 patients treated with highdose chemotherapy and autologous peripheral blood stem cell transplantation. Bone Marrow Transpl. 1998;23(1):27–33.
- Jones GT, Abramson N. Gastrointestinal necrosis in acute leukemia: a complication of induction therapy. Cancer Invest. 1983;1(4):315–20.
- Gorschlüter M, Mey U, Strehl J, Ziske C, Schepke M, Schmidt-Wolf IGH, et al. Neutropenic enterocolitis in adults: Systematic analysis of evidence quality. Eur J Haematol. 2005;75(1):1–13. https://doi.org/10.1111/j.1600-0609.2005.00442.x.
- Ullery BW, Pieracci FM, Rodney JRM, Barie PS. Neutropenic enterocolitis. Surg Infect. 2009;10(3):307–14.
- Gomez L, Martino R, Rolston KV. Neutropenic enterocolitis: spectrum of the disease and comparison of definite and possible cases. Clin Infect Dis. 1998;27(4):695–9.
- Cunningham SC, Fakhry K, Bass BL, Napolitano LM. Neutropenic enterocolitis in adults: case series and review of the literature. Digest Dis Sci. 2005;50(2):215–20.
- Gorbach SL. Neutropenic enterocolitis. Clin Infect Dis. 1998;27(4):700–1.
- Davila ML. Neutropenic enterocolitis. Curr Treat Options Gastroenterol. 2006;9(3):249–55.
- Abu-Sbeih H, Ali FS, Chen E, Mallepally N, Luo W, Lu Y, et al. Neutropenic enterocolitis. Dis Colon Rectum. 2020;63(3):381–8.
- Sachak T, Arnold MA, Naini BV, Graham RP, Shah SS, Cruise M, et al. Neutropenic enterocolitis: new insights into a deadly entity. Am J Surg Pathol. 2015;39(12):1635–42. https://doi.org/10.1097/ PAS.000000000000517.
- Kouroussis C, Samonis G, Androulakis N, Souglakos J, Voloudaki A, Dimopoulos M-A, et al. Successful conservative treatment of neutropenic enterocolitis complicating taxane-based chemotherapy. Am J Clin Oncol Cancer Clin Trial. 2000;23(3):309–13.
- Amromin GD, Solomon RD. Necrotizing enteropathy. JAMA. 1962;182(1):23.
- Wade DS, Nava HR, Douglass HO. Neutropenic enterocolitis. Clinical diagnosis and treatment. Cancer. 1992;69(1):17–23.

- Song HK, Kreisel D, Canter R, Krupnick AS, Stadtmauer EA, Buzby G. Changing presentation and management of neutropenic enterocolitis. Arch Surg. 1998;133(9):979–82.
- Hsu TF, Huang HH, Yen DH, Kao WF, Chen JD, Wang LM, Lee CH. ED presentation of neutropenic enterocolitis in adult patients with acute leukemia. Am J Emerg Med. 2004;22(4):276–9.
- Rodrigues FG, Dasilva G, Wexner SD. Neutropenic enterocolitis. World J Gastroenterol. 2017;23(1):42.
- Bibbo C, Barbieri RA, Deitch EA, Brolin RE. Neutropenic enterocolitis in a trauma patient during antibiotic therapy for osteomyelitis. J Trauma. 2000;49(4):760–3.
- Cutrona AF, Blinkhorn RJ, Crass J, Spagnuolo PJ. Probable neutropenic enterocolitis in patients with aids. Rev Infect Dis. 1991;13(5):828–31.
- Mullassery D, Bader A, Battersby AJ, Mohammad Z, Jones ELL, Parmar C, et al. Diagnosis, incidence, and outcomes of suspected typhlitis in oncology patients—experience in a tertiary pediatric surgical center in the United Kingdom. J Pediatr Surg. 2009;44(2):381–5.
- 22. Badgwell BD, Cormier JN, Wray CJ, Borthakur G, Qiao W, Rolston KV, et al. Challenges in surgical management of abdominal pain in the neutropenic cancer patient. Ann Surg. 2008;248(1):104–9.
- Geisler JP, Schraith DF, Manahan KJ, Sorosky JI. Gemcitabine associated vasculitis leading to necrotizing enterocolitis and death in women undergoing primary treatment for epithelial ovarian/peritoneal cancer. Gynecol Oncol. 2004;92(2):705–7.
- Kronawitter U, Kemeny NE, Blumgart L. Neutropenic enterocolitis in a patient with colorectal carcinoma: unusual course after treatment with 5-fluorouracil and leucovorin--a case report. Cancer. 1997;80(4):656–60.
- User İR, Akbayram S, Özokutan BH. Clinical presentation, prognostic factors, and outcome in neutropenic enteropathy of childhood leukemia. J Pediatr Hematol Oncol. 2018;40(3):216–20.
- Bow EJ, Meddings JB. Intestinal mucosal dysfunction and infection during remission-induction therapy for acute myeloid leukaemia. Leukemia. 2006;20(12):2087–92.
- 27. Cartoni C, Dragoni F, Micozzi A, Pescarmona E, Mecarocci S, Chirletti P, et al. Neutropenic enterocolitis in patients with acute leukemia: prognostic significance of bowel wall thickening detected by ultrasonography. J Clin Oncol. 2001;19(3):756–61.
- Mehdi I, Bahrani BA. Chemotherapy-induced neutropenic necrotizing enterocolitis: a review. J Pak Med Assoc. 2012;62(7):718–23.
- Kaste SC, Flynn PM, Furman WL. Acute lymphoblastic leukemia presenting with typhlitis. Med Pediatr Oncol. 1997;28(3):209–12.
- de Brito D, Barton E, Spears KL, Cranmer HH, Karp SJ, Anglin D, et al. Acute right lower quadrant pain in a patient with leukemia. Ann Emerg Med. 1998;32(1):98–101.
- Prolla JC, Kirsner JB. The gastrointestinal lesions and complications of the leukemias. Ann Intern Med. 1964;61:1084–103.
- 32. Yuen KY, Woo PC, Liang RH, Chiu EK, Chen FF, Wong SS, et al. Clinical significance of alimentary tract microbes in bone marrow transplant recipients. Diagn Microbiol Infect Dis. 1998;30(2):75–81.
- 33. Gorschlüter M, Mey U, Strehl J, Schmitz V, Rabe C, Pauls K, et al. Invasive fungal infections in neutropenic enterocolitis: a systematic analysis of pathogens, incidence, treatment and mortality in adult patients. BMC Infect Dis. 2006;6(1):35.
- Mower WJ, Hawkins JA, Nelson EW. Neutropenic enterocolitis in adults with acute leukemia. Arch Surg. 1986;121(5):571–4.
- Rolston KV, Bodey GP, Safdar A. Polymicrobial infection in patients with cancer: an underappreciated and underreported entity. Clin Infect Dis. 2007;45(2):228–33.

- 36. Rizzatti M, Brandalise SR, Azevedo ACD, Pinheiro VRP, Aguiar SDS. Neutropenic enterocolitis in children and young adults with cancer: prognostic value of clinical and image findings. Pediatr Hemat Oncol. 2010;27(6):462–70.
- McCarville MB, Adelman CS, Li C, Xiong X, Furman WL, Razzouk BI, et al. Typhlitis in childhood cancer. Cancer. 2005;104(2):380–7.
- Alioglu B, Avci Z, Ozcay F, Arda S, Ozbek N. Neutropenic enterocolitis in children with acute leukemia or aplastic anemia. Int J Hematol. 2007;86(4):364–8.
- Sloas MM, Flynn PM, Kaste SC, Patrick CC. Typhlitis in children with cancer: a 30-year experience. Clin Infect Dis. 1993;17(3):484–90.
- 40. Gorschlüter M, Marklein G, Höfling K, Clarenbach R, Baumgartner S, Hahn C, et al. Abdominal infections in patients with acute leukaemia: a prospective study applying ultrasonography and microbiology. Br J Haematol. 2002;117(2):351–8.
- Kirkpatrick IDC, Greenberg HM. Gastrointestinal complications in the neutropenic patient: characterization and differentiation with abdominal CT. Radiology. 2003;226(3):668–74.
- Teefey SA, Montana MA, Goldfogel GA, Shuman WP. Sonographic diagnosis of neutropenic typhlitis. Am J Roentgenol. 1987;149(4):731–3.
- Dietrich CF, Hermann S, Klein S, Braden B. Sonographic signs of neutropenic enterocolitis. World J Gastroenterol. 2006;12(9):1397–402.
- 44. Pugliese N, Salvatore P, Iula DV, Catania MR, Chiurazzi F, Pepa RD, et al. Ultrasonography-driven combination antibiotic therapy with tigecycline significantly increases survival among patients with neutropenic enterocolitis following cytarabine-containing chemotherapy for the remission induction of acute myeloid leukemia. Cancer Med. 2017;6(7):1500–11.
- Picardi M, Camera A, Pane F, Rotoli B. Improved management of neutropenic enterocolitis using early ultrasound scan and vigorous medical treatment. Clin Infect Dis. 2007;45(3):403–4.
- 46. Hughes WT, Armstrong D, Bodey GP, Bow EJ, Brown AE, Calandra T, et al. 2002 guidelines for the use of antimicrobial agents in neutropenic patients with cancer. Clin Infect Dis. 2002;34(6):730–51.
- Williams N, Scott ADN. Neutropenic colitis: a continuing surgical challenge. Brit J Surg. 1997;84(9):1200–5.
- Katz JA, Mahoney DH, Fernbach DJ, Wagner ML, Gresik MV. Typhlitis. An 18-year experience and postmortem review. Cancer. 1990;65(4):1041–7.
- Shamberger RC, Weinstein HJ, Delorey MJ, Levey RH. The medical and surgical management of typhlitis in children with acute nonlymphocytic (myelogenous) leukemia. Cancer. 1986;57(3):603–9.
- Cloutier RL. Neutropenic enterocolitis. Emerg Med Clin N Am. 2009;27(3):415–22. https://doi.org/10.1016/j.emc.2009.04.002.
- Dexheimer JW, Kurowski BG, Anders SH, McClanahan N, Wade SL, Babcock L. Usability evaluation of the SMART application for youth with mTBI. Int J Med Inform. 2017;97:163–70.
- 52. Saillard C, Zafrani L, Darmon M, Bisbal M, Chow-Chine L, Sannini A, et al. The prognostic impact of abdominal surgery in cancer patients with neutropenic enterocolitis: a systematic review and meta-analysis, on behalf the Groupe de Recherche en Réanimation Respiratoire du patient d'Onco-Hématologie (GRRR-OH). Ann Intensive Care. 2018;8(1):47.
- 53. Smith TJ, Bohlke K, Lyman GH, Carson KR, Crawford J, Cross SJ, et al. Recommendations for the use of WBC growth factors: American Society of Clinical Oncology Clinical Practice Guideline Update. J Clin Oncol. 2015;33(28):3199–212.



Sickle Cell Disease

Harsh P. Sule, Novneet N. Sahu, and Bernard L. Lopez

Case Study

A 25-year-old female with history of sickle cell disease (SCD) presents to the emergency department reporting 10/10 dull, aching pain in her back and hips. She has been taking her home regimen of morphine around the clock without relief. She took her last dose of medication just prior to arrival. She states that this is typical of prior episodes where she has needed care in the emergency department. Her vital signs are temperature of 37.2 °C, heart rate of 80 beats per minute, blood pressure of 112/72, and oxygen saturation of 99% on room air. She has a normal exam, and you do not suspect dehydration, organ failure, or neurological complications. Her old records are reviewed, and she has been previously given higher doses of morphine in the emergency department with relief. You suspect an acute painful episode associated with vaso-occlusive crisis and provide her repeat doses of pain medication. Her symptoms improve after the third dose of pain medication and she feels well. You discharge her home to continue her outpatient regimen and she will follow up with her hematologist.

Introduction

Emergency physicians (EPs) practicing in North America must be familiar with SCD and its complications. SCD is the most common inherited blood disorder in the United States [1]: 1 in 13 African American newborns has sickle cell trait, while 1 in 365 African Americans has the disease [2]. In SCD, hemoglobin molecules have a propensity to aggregate

H. P. Sule · N. N. Sahu

into rigid polymers, particularly under conditions of low oxygen tension, resulting in the characteristic sickle-shaped erythrocytes that cause vaso-occlusion and ischemia. The clinical hallmark of SCD is episodes of acute pain, and this is by far the most common reason for emergency department (ED) visits and inpatient admissions by SCD patients [3–6]. In addition to managing these acute pain episodes, the EP must be alert to other manifestations, complications, and comorbidities of the disease, some of which carry significant risk for morbidity and mortality.

This chapter describes the clinical presentations and management of SCD in the ED while delving into key decisions and current controversies.

Pathophysiology

SCD was first described in the Western medical literature by James B. Herrick in 1910 [7]. In 1949 James V. Neel described the pattern of inheritance, with individuals who were heterozygous for the responsible gene having sickle cell trait (SCT) and homozygous individuals having SCD [8]. SCD was the first human anemia defined at the amino acid level [9].

Human hemoglobin (Hb) molecules are typically tetramers comprising four subunit proteins (two α and two β subunits) [10]. The exact composition of the peptide chains determines the specific shape into which the molecule can fold. Hemoglobin S (HbS) is the result of a glutamic acid to valine substitution at the $\beta 6$ amino acid position [11, 12]. The result is polymerization due to a hydrophobic interaction between the altered, deoxygenated molecule and other hemoglobin molecules [11]. As a consequence, there are a change in the shape and reduction in the critical ability of erythrocytes to deform [11, 12]. While initially thought that the ensuing change in flow characteristics and erythrocyte aggregation alone caused vaso-occlusion, the root cause is multi-factorial. Initial endothelial activation with increased adhesion of erythrocytes and leukocytes is followed by

© Springer Nature Switzerland AG 2021

Department of Emergency Medicine, Rutgers New Jersey Medical School, Newark, NJ, USA

B. L. Lopez (⊠) Department of Emergency Medicine, Sidney Kimmel Medical College, Thomas Jefferson University & Hospitals, Philadelphia, PA, USA e-mail: bernard.lopez@jefferson.edu

K. H. Todd et al. (eds.), Oncologic Emergency Medicine, https://doi.org/10.1007/978-3-030-67123-5_41

formation of heterocellular aggregates that then physically result in occlusion and local hypoxia [11]. This triggers a vicious cycle of increased HbS formation due to hypoxia, presence of inflammatory mediators, free radicals, and reperfusion injury. Hemoglobin also binds nitric oxide (NO), a potent vasodilator, and releases it with oxygen [10]. Ineffective binding and release of NO along with hemolysis and erythrocyte lysis further reduce NO production and result in persistent tissue hypoxia [10, 11, 13]. Finally, erythrocytes are more likely to sickle and become rigid the more dehydrated they get. This is in large part due to changes in cation homeostasis - specifically, increased potassium and water efflux mediated by potassium-chloride co-transport and Gardos channels (Ca⁺⁺-dependent K⁺ channel) [11, 14]. Figure 41.1 [12] displays the pathophysiology of sickle cell disease.

Sickle cell disease is commonly represented by the primary HbSS genotype. There are also five other genotypes that are associated with varying clinical severity, and all have a majority of their hemoglobin as HbS [13]. HbSS, commonly referred to as sickle cell SS disease, is the most severe clinically, with the heterozygous HbS/ β° thalassemia genotype being similarly severe. HbSC has intermediate severity, HbS/ β^{+} has mild to moderate severity, and HbS/HPFH (hereditary persistence of HbF) and HbS/HbE demonstrate mild to no symptoms [14]. There exist several other genotypes that are exceedingly rare but do cause disease of varying severity. Those with sickle cell trait (heterozygous with HbA) have hemoglobin that is majority HbA [13]. Table 41.1 [15] displays the genotypes and phenotypes of different sickling disorders.



Fig. 41.1 Pathophysiology of sickle cell disease. (From Rees et al. [12], with permission of The Lancet and Elsevier)

 Table 41.1 Genotypes and characteristics of different sickling disorders

HbA (%)	HbS (%)	HbC (%)	HbF (%)	HbA ₂ (%)	Clinical course
95– 98	0	0	<1	<3.5	-
55– 65	30– 40	0	<1	<3.5	Benign
55– 65	0	30– 40%	<1	<3.5	Benign
90– 95	0	0	1–3	>3.5	Benign
Disease conditions					
0	80– 95	0	5–15	<3.5	Severe
0	50– 55	40– 45%	<3	<3.5	Moderate
0	80– 90	0	5-15	>3.5	Severe
10– 25	70– 80	0	<3	>3.5	Mild
0	50– 60	0	Variable	<3.5	Variable
	HbA (%) 95- 98 55- 65 55- 65 90- 95 0 0 0 0 10- 25 0	HbA HbS (%) (%) 95- 0 98 30- 55- 40 55- 0 65 0 90- 0 95- 0 90- 0 90- 0 90- 0 90- 0 90- 0 90- 0 90- 10 90- 10- 25- 80 00- 50- 60 80-	HbA (%) HbC (%) HbC (%) 95- 0 (%) 95- 0 0 98 $30-$ 0 55- 40 0 55- 0 $30-$ 65 40 0 90- 0 0 90- 0 0 90- 0 0 90- 0 0 90- 0 0 90- 0 0 90- 0 0 90- 0 0 90- 0 0 90- 0 0 90- 0 0 90- 0 0 90- 0 0 90- 0 0 90- 0 0 90- 0 0 90- 0 0 90- 0 0 90- 0 0<	HbA (%)HbS (%)HbC (%)HbF (%)95- 9800<1	$\begin{array}{c ccccc} \text{HbA} & \text{HbS} & \text{HbC} & \text{HbF} & \text{HbA}_2 \\ (\%) & (\%) & (\%) & (\%) & (\%) \\ 95- & (\%) & (\%) & (\%) & (\%) \\ 95- & (\%) & (\%) & (\%) & (\%) \\ 95- & (\%) & (\%) & (\%) & (\%) \\ 55- & (\%) & (\%) & (\%) & (\%) \\ 55- & (\%) & (\%) & (\%) & (\%) \\ 55- & (\%) & (\%) & (\%) & (\%) \\ 55- & (\%) & (\%) & (\%) & (\%) \\ 65- & (\%) & (\%) & (\%) & (\%) \\ 60- & (\%) & (\%) & (\%) & (\%) \\ 60- & (\%) & (\%) & (\%) & (\%) \\ 10- & (\%) & (\%) & (\%) & (\%) \\ 10- & (\%) & (\%) & (\%) & (\%) \\ 10- & (\%) & (\%) & (\%) & (\%) \\ 10- & (\%) & (\%) & (\%) & (\%) \\ 10- & (\%) & (\%) & (\%) & (\%) & (\%) \\ 10- & (\%) & (\%) & (\%) & (\%) & (\%) \\ 10- & (\%) & (\%) & (\%) & (\%) & (\%) \\ 10- & (\%) & (\%) & (\%) & (\%) & (\%) \\ 10- & (\%) & (\%) & (\%) & (\%) & (\%) \\ 10- & (\%) & (\%) & (\%) & (\%) & (\%) & (\%) \\ 10- & (\%) & (\%) & (\%) & (\%) & (\%) & (\%) \\ 10- & (\%) & (\%) & (\%) & (\%) & (\%) & (\%) \\ 10- & (\%) & (\%) & (\%) & (\%) & (\%) & (\%) & (\%) \\ 10- & (\%) & (\%) & (\%) & (\%) & (\%) & (\%) & (\%) & (\%) & (\%) & (\%) \\ 10- & (\%) & $

Adapted from Ware et al. [15], with permission Elsevier

Epidemiology

Based on geographical distribution and genetic studies, the sickle cell gene is thought to provide protection against malaria infection by *Plasmodium falciparum* in heterozygotes [16]. Despite the length of time since this association was first proposed, the exact cellular mechanisms for this protection remain unclear [17]. Geographic distribution of HbS genotypes is based on the presence of malaria in a region and ensuing migration trends. While SCD is most common among people of African descent, it is also found among people of South Asian, Middle Eastern, and Mediterranean descent [18]. There are four African haplotypes and one Arab-Indian haplotype [12].

In the United States, prevalence of SCD is highest in the African American population (1 in 365 Black or African American) with the average age of death in all patients with SCD being 39 years [19]. With advances in medical care, survival to the age of 18 is now 93.9% with a mortality rate in pediatric patients of 0.52 per 100 patient years [20].

Clinical Presentations in the ED

Table 41.2 summarizes the various acute and chronic clinical presentations of sickle cell anemia [21]. Figure 41.2 shows the biochemical targets of potential treatments for SCD [21].

Pain in Sickle Cell Disease

While EPs most frequently encounter acute pain associated with SCD, they must become familiar with other types of pain that occur in patients with SCD (Table 41.3) [22]. The EP faces three important challenges. First, the EP must recognize other causes of pain that may masquerade as vasoocclusive crisis (VOC) pain, in some instances representing dangerous conditions such as acute chest syndrome (abbreviated here as AChS to avoid confusion with acute coronary syndrome). However, adult patients with SCD can reasonably differentiate their VOC-related pain from other causes [23]. Second, the EP must be alert to other dangerous conditions that are not painful and are masked by the patient's preoccupation with their acute pain. Third, the EP must

Table 41.2 Clinical manifestations of sickle cell ane

Acı

	a 1 1	
	Complication	Characteristics
ite	Acute pain episodes (vaso-occlusive events)	Most common complication of SCD. Severity varies and may be manageable at home or require hospitalization
	Acute chest syndrome	Development of chest pain accompanied by fever, respiratory symptoms, and a chest X-ray with a new pulmonary opacity; associated with hypoxemia in severe cases
	Stroke	Both ischemic and hemorrhagic stokes may occur in adult SCD
	Thrombosis	Can occur in adult SCD, particularly during pregnancy and post-partum
	Liver complications	Acute pain in the right upper quadrant with jaundice requires workup to distinguish between acute cholecystitis, acute viral hepatitis, hepatic sequestration, and sickle cell intra-hepatic cholestasis
	Infections	Most frequently pneumonia, osteomyelitis, and urinary infections; may progress to or present primarily as sepsis
	Priapism	Compartment syndrome of the penis causing "stuttering" (short-lived, intermittent) or prolonged (lasting over 4 h) painful penile erections; can cause permanent erectile dysfunction
	Aplastic crisis	Exacerbated anemia accompanied by reticulocytopenia, usually caused by parvovirus B19 infection

(continued)

Table 41.2 (continued)

	Complication	Characteristics
Chronic	Hemolytic anemia	Normocytic, normochromic anemia; hemoglobin levels may vary between 6 and 10 g/dL, accompanied by reticulocytosis
	Functional asplenia	Splenic dysfunction and eventual autosplenectomy secondary to the splenic infarction that usually occurs during childhood
	Avascular necrosis	Can affect hip(s) or shoulder(s) causing early osteoarthritis and chronic pain
	Osteopenia and osteoporosis	Reduction of bone mass density occurs earlier than in the general population; is progressive and associated with hemolysis in SCD
	Pulmonary arterial hypertension	Exertional dyspnea or fatigue with chronic oxygen desaturation, caused by pulmonary artery lumen restriction and wall stiffening, linked to hemolysis, and associated with poor prognosis
	Gallstones/	Caused by augmented heme
	Retinopathy	Proliferative retinopathy is relatively frequent, especially in HbSC disease and may cause blindness
	Nephropathy	Hyperfiltration and hyposthenuria occur early; incidence of microalbuminuria increases with patient age and can result in end-stage kidney disease
	Heart disease	Includes diastolic dysfunction with increased mortality, overt heart failure, and under- recognized acute myocardial infarction
	Leg ulcers	Development of ulcers in the maleolar and distal leg skin can be a recurring complication. These ulcers are painful, disfiguring, and are difficult to heal
	Neurological complications	Neurocognitive impairment is frequent, particularly in patients with previous stroke. Moyamoya syndrome may also occur, with proliferation of intracerebral blood vessels caused by stenosis or occlusion of cerebral arteries, increasing risk for acute cerebrovascular events
		cerebrovascular events



Fig. 41.2 Biochemical aspects of the management of sickle cell disease. (From Costa et al. [21], with permission Springer Nature)

Table 41.3 Types of pain in sickle cell disease

ACS acute chest syndrome, AVN avascular necrosis, NMDA N-methyl-D-aspartate, VOC vaso-occlusive crisis

From Costa et al. [21], with permission Springer Nature

promptly initiate analgesia and other therapy for the acute painful episode.

Chronic Pain in Sickle Cell Disease The Pain in Sickle Cell Epidemiology Study (PiSCES) found that in adult patients pain was reported in 56% of patient-days, with 29.3% of patients reporting pain on more than 95% of days logged [24]. Furthermore, this study identified that only 3.5% of patient-days were associated with utilization of healthcare services. These numbers are important to recognize in terms of reducing the biases that healthcare providers often harbor toward patients with SCD. The Analgesic, Anesthetic, and Addiction Clinical Trial Translations Innovations Opportunities and Networks-American Pain Society Pain Taxonomy (AAPT) initiative has developed diagnostic criteria for chronic sickle cell disease pain involving five dimensions [23]. These should be reviewed by providers in ED observation units and day units for sickle patients to gain familiarity with the spectrum of pain SCD patients experience. Table 41.4 outlines the AAPT criteria [23].

 Table 41.4
 American Pain Society Pain Taxonomy diagnostic criteria

 for chronic pain associated with sickle cell disease (SCD)

Dimension 1: Core diagnostic criteria

- 1. Diagnosis of SCD confirmed by laboratory testing
- 2. Reports of ongoing pain present on most days over the past
- 6 months either in a single location or in multiple locations 3. Must display *at least 1* sign:
- Palpation of the region of reported pain elicits focal pain or tenderness
- Movement of the region of reported pain elicits focal pain
- Decreased range of motion or weakness in the region of reported pain
- Evidence of skin ulcer in the region of reported pain
- Evidence of hepatobiliary or splenic imaging abnormalities (e.g., splenic infarct, chronic pancreatitis) consistent with the region of reported pain
- Evidence of imaging abnormalities consistent with bone infarction or avascular necrosis in the region of reported pain
- 4. There is no other diagnosis that better explains the signs and symptoms

Chronic SCD pain diagnostic modifiers:

We propose three diagnostic modifiers to indicate subtypes of chronic SCD pain

- Chronic SCD pain without contributory disease complications is used if there is no evidence of contributory SCD complications on the basis of either clinical signs (e.g., presence of leg ulcers) or test results (e.g., imaging abnormalities)
- Chronic SCD pain with contributory disease complications should be used if there is evidence of contributory SCD complications on the basis of clinical signs or test results
- 3. Chronic SCD pain with mixed pain types should be used if there is evidence of contributory SCD complications (e.g., avascular necrosis) on the basis of clinical signs or test results and there is pain also occurring in unrelated sites (e.g., arms, back, chest, or abdominal pain)

From Dampier et al. [23], with permission Elsevier

Vaso-occlusive Crisis and Acute Painful Episodes Vasoocclusive crisis (VOC) is the most common ED presentation of SCD. Patients with HbSS or HbS β 0 experience frequent VOCs, while those with sickle cell trait do not experience typical VOCs [25]. Microvascular occlusion (the cardinal pathophysiologic cause of acute pain) leads to ischemia and hypoxia. This is followed by tissue and vascular damage and inflammation, with release of inflammatory mediators, all of which activate nociceptors [22, 26]. Reperfusion intensifies the inflammation and resultant pain [3, 4, 6]. Patients experiencing frequent pain crises (more than three per year) are at higher risk for complications including a higher mortality rate [23].

Classically, acute pain from VOC is described in the back or extremities, although it may occur elsewhere. It may be migratory and is usually continuous and progressive. VOC pain occurs less frequently elsewhere, such as in the chest, where it may create diagnostic overlap with acute chest syndrome and other causes of chest pain. Pain from VOC should be distinguished from two other patterns of pain experienced by patients with SCD: acute flare of chronic pain and neuropathic pain. Generally, the major focus for the EP is upon management of acute painful episodes caused by VOC.

There are no reliable signs or tests to indicate the presence or absence of VOC or pain associated with VOC. Patients' self-reported pain scores do not reliably correlate with changes in vital signs, such as tachycardia, hypertension, or tachypnea, that may be associated with pain in other clinical contexts. Hemoglobin, hematocrit, and reticulocyte measurements do not serve as markers for pain; neither do some more recent tests such as inflammatory cytokines [26]. Further, patients may exhibit behavior which EPs consider inconsistent with pain, including walking, engaging in conversations, or having a calm appearance while still reporting high levels of pain [3, 5, 27]. Laboratory tests should be ordered for patients who are being admitted and for other indications that may be considered.

Opioids have been the mainstay of analgesia for acute painful episodes over the past century. Meperidine was commonly used in the past but is no longer a commonly available or utilized medication because it is metabolized to normeperidine and has been linked to emotional and behavioral changes as well as seizures due to renal excretion and a long half-life [28–30]. More recently morphine, fentanyl, and hydromorphone became the medications of choice in EDs. However, with the recent opioid crisis and related stigma and addiction concerns, there is an emphasis on using these in an incremental and targeted manner.

Contemporary management of acute painful episodes is outlined in Table 41.5. Key goals include prioritization of these patients in triage, early and adequate analgesia, and frequent repeat dosing until pain is controlled [3, 22, 25–31].
 Table 41.5
 Emergency department (ED) management of sickle cell disease-related pain

Triage	ca
Triage as high priority ESI level 2	pa
Assess for complications and administer first dose of analgesic	tic
within 30 min of triage or within 60 min of registration	uc
Route of administration	
Oral route preferred for initial dose	
Intranasal can be an alternative for appropriate medication	A
Intravenous should be used for second-line therapy	
Avoid intramuscular route due to unpredictable pharmacokinetics	Δ.
Medications	A
Opioid naïve: NSAIDs such as ibuprofen and ketorolac for	of
mild-moderate pain, with oral immediate-release morphine or	ni
intranasal fentanyl for severe pain	ta
Opioid tolerant: Oral immediate-release morphine based on home	A
dose, with intranasal fentanyl as an alternative	in
Intravenous morphine if unable to tolerate or failed oral	1
administration	la
Pain/sub-dissociative dose ketamine for persistent/uncontrolled	re
pain	35
Avoid meperidine and hydromorphone	fe
Reassessment	0
Within 30 min	4:
Use capnography if available before initiating additional or higher	d1
doses of morphine	sy
Consider admission or placement in observation unit if pain is not	
controlled within 90 min with adequate dosing	
Aajuncts	Та
Ondansetron for nausea or vomiting	of
first does only)	
A cetaminonhen as an adjunct for pain relief	C
Consider non pharmacologic approaches such as heat	S
acupuncture etc	
Other therapies	
Introvenous fluids: routing fluids not indicated in the ED if	
euvolemic – holus only if hypovolemic	
Oxygen: only if hypoxic (oxygen saturation less than 92%)	
Nitric oxide: vasodilatory effect, but limited data on benefits with	
potential risks	
Hydroxyurea: increased HbF and reduces frequency of VOC	
j j indicased flot and reduces neglectof of 100	

ESI Emergency Severity Index, *NSAID* non-steroidal anti-inflammatory drug, *VOC* vaso-occlusive crisis

Avoid using the intramuscular route and administer fluids only to patients who are hypovolemic.

Patients with acute painful episodes may be discharged if their pain is adequately controlled and they expect, based on past individual experience, that it will remain adequately controlled with ongoing use of their outpatient analgesic regimen. There is no evidence-based fixed number of doses of opioid medication established as a threshold for admission decisions, and EPs should judge the necessity of admission on clinical grounds, balancing the benefits of pain control with parenteral opioids and adjuvants against the well-known risks of hospitalization.

Day clinics are a complement and alternative to traditional inpatient therapy for acute painful episodes [5, 31]. Day clinics are less resource intensive than EDs, with less associated pres-

sure on length of stay and more room to achieve pain control and discharge to home without repeat presentation. Day clinics can additionally complement outpatient hematology clinics as part of a medical home for patients with SCD, providing a multidisciplinary approach beyond acute pain management.

Acute Chest Syndrome

Acute chest syndrome (AChS) is defined as the appearance of a new pulmonary infiltrate on chest radiography accompanied by a fever and respiratory symptoms including cough, tachypnea, and chest pain [32, 33]. It is hypothesized that AChS is the result of hypoxia and an inflammatory mediatorinduced increase in adhesion of the pulmonary microvasculature to sickled erythrocytes. This is coupled with a reduction in NO that would normally counteract this [34, 35]. The most common symptoms in patients with AChS are fever (80%) and cough (62–74%), with rales being the most common finding on physical exam (48–76%) [33]. Table 41.6 displays clinical and laboratory characteristics of acute chest syndrome [33].

 Table 41.6
 Select clinical and laboratory characteristics at diagnosis

 of acute chest syndrome
 Image: Clinical and laboratory characteristics at diagnosis

	CSSCD	MACSSD	
Symptoms at diagnosis, %			
Fever	80	80	
Cough	74	62	
Chest pain	57	44	
Tachypnea	28	45	
Pain in arms and legs	NR	37	
Pain in ribs and sternum	NR	21	
Pain in abdomen	NR	35	
Reactive airway disease	NR	13	
Neurologic dysfunction	NR	4	
Physical findings/lab investigations at diagnosis			
Respiratory rate >30/min, %	18 (>40/min)	67	
Wheeze, %	10	26	
Rales, %	48	76	
Normal auscultation, %	35	NR	
Mean Hb (g/dL) in HbSS patients	7.9	7.7	
Mean leukocyte count (103/mm ³)	21.1	23	
Mean O2 saturation	NR	92	
Mean Pa O2 (mm Hg)	71	70	
Effusion (% adults)	21	36	
Effusion (% children <10 y)	3	34	
Bacteremia, %	3.5	NR	
Admitted for reasons other than			
Symptoms of ACS, %	42	48	
Duration hospital stay for adults (d)	9	12.8	
Duration hospital stay for child (d)	5.4	9.7	

From Stuart and Setty [33], with permission Wolters Kluwer CSSCD Cooperative Study of Sickle Cell Disease, MACSS Multi-Center Acute Chest Syndrome Study Group, ACS acute chest syn-

drome, NR not recorded

While a significant portion (45.7%) of cases of AChS are due to unknown etiologies, 30% of AChS cases yield a documented infection, and about 9% are attributed to a pulmonary fat embolism (which in turn could be due to infection) and 16% to pulmonary infarction (not caused by infection or fat embolism) [35]. Hypoventilation due to two common clinical scenarios puts the patient at risk for AChS: splinting (pain) and/or opioid analgesic side effects (somnolence) [33]. Pulmonary edema caused by intravenous hydration may lead to AChS, but the data is not conclusive. Pulmonary thrombosis occurs in about 17% of cases with AChS, but it is unclear if it is a cause of AChS [36]. Additionally, the use of clinical decision rules such as the revised Geneva score and d-dimer testing have not been found to be useful in patients with AChS [36].

Initial management of AChS includes empiric antibiotics (cephalosporin and macrolide - to ensure coverage of chlamydia and mycoplasma); however, antibiotics do not appear to have a significant effect on patient outcomes [25]. In a single, small randomized controlled trial, dexamethasone use was shown to have a short-term benefit on hospital length of stay, need for transfusions, number of opioid doses, and clinical condition [37]; however, with longer follow-up, results were inconsistent. There is wide variability in dexamethasone use between hospitals, and in the absence of larger randomized clinical trials, it is difficult to determine its efficacy with any precision [38]. Incentive spirometry plays a significant role in the prophylaxis and treatment of AChS [25]. Pain medications and intravenous hydration should be administered judiciously due to their risk of worsening the condition.

Blood transfusion has been used effectively in the United States for treatment of AChS, but a significant improvement in mortality rate has not been noted when compared to Europe, where it is not used as routinely [35]. Additionally, substantial increases in hemoglobin levels can increase viscosity, thereby raising the risk of complications. Exchange transfusion may be considered in patients with a baseline "high" hemoglobin or those with "multilobe involvement, persistent or worsening hypoxemia, neurologic abnormalities, or multi-organ failure" but has also had mixed results [25, 35, 39, 40]. Both transfusions retain a "Strong Recommendation" from the National Heart, Lung, and Blood Institute's Expert Panel but with low-quality evidence [25].

The role of anti-coagulation in AChS patients with pulmonary thrombosis remains unclear, but EPs would be hardpressed to avoid anti-coagulation if CT findings are consistent with a pulmonary artery occlusion. Nitric oxide inhalation and steroids may also be used in the management of AChS, but at present, these therapies should be considered experimental [35]. 535

Disposition of these patients should be dictated by their clinical status. While increasing age is associated with worse outcomes (adults generally do worse than pediatric patients) and up to 3% of patients with AChS die, absolute predictors of bad outcomes are difficult to define. The best predictors of bad outcome are extensive lung involvement on x-ray, a platelet count of less than 199,000/mm³, and cardiac disease as a comorbidity [35]. All patients with AChS should initially be admitted to a monitored setting (not necessarily an intensive care unit).

Fever and Infection

Patients with SCD are at high risk for infections with encapsulated organisms due to functional asplenia, as well as a functionally immunocompromised state (increased bone marrow turnover and altered complement activation) [4]. Penicillin prophylaxis in children and widespread use of the pneumococcal vaccine have made tremendous headway in reducing the incidence of bacterial infections and sepsis [25]. However, EPs must remain aggressive in working up any SCD patient presenting with a fever while keeping in mind that their reduced ability to mount an inflammatory response may limit leukocytosis or even an elevated temperature.

A low threshold should be employed to order a complete blood count (CBC), chest x-ray, and urinalysis at minimum. If no source if identified, consideration should be given to obtaining blood and urine cultures, performing a lumbar puncture, evaluating for potential arthrocentesis, and initiating empiric antibiotics. Workup and treatment should be even more aggressive in pediatric patients since the risk of overwhelming *Streptococcus pneumoniae* sepsis is high [25, 41].

Admission to the hospital should be strongly considered for ill-appearing patients with SCD and a temperature ≥ 103.1 °F (39.5 °C) [25]. Outpatient management with administration of parenteral long-acting antibiotics may be considered in select patients with reliable follow-up (within 24 h) as long as they defervesce, have never been septic, and have a normal chest radiograph, oxygen saturation, baseline white blood cell (WBC) count, platelet count, and hemoglobin.

Pulmonary Hypertension

Pulmonary hypertension (PHTN) has an incidence of 6-10% and a mortality rate of 2-5% [13]. PHTN is thought to be due in large part to changes in medial smooth muscle and endothelial cells. The key clinical finding is reduced exercise

capacity (45% of patients are New York Heart Association class III or IV) [13]. Diagnostic findings include an elevated N-terminal pro-brain natriuretic peptide (NT-proBNP), an elevated tricuspid valve regurgitant jet velocity on echocardiography, and elevated pulmonary pressures on right heart catheterization.

Currently there is no approved and effective therapy to manage PHTN in SCD patients. Sildenafil, used in non-SCD-related PHTN, has been studied but was found to increase acute pain episodes [13].

Cerebrovascular Accident

While strokes are unusual entities in the general pediatric population, they are relatively frequent in children with SCD and more common in children than adults with SCD [12, 13]. Cerebrovascular accidents (CVA) can occur in children as young as 2 years of age, with 11% of patients with SCD having a stroke by 20 years of age [12]. However, silent cerebral infarcts (SCI) associated with small-vessel disease are more common than overt strokes with 34% of SCD patients having evidence of SCI by age 14 [13]. Symptomatic patients exhibit vasculopathy primarily in the distribution of mid- to large-sized arteries (distal internal carotid and middle cerebral arteries) which is triggered by anemia, leukocytosis, hypoxia, and impaired regulation of blood flow [12, 13].

The Stroke Prevention Trial in Sickle Cell Anemia (STOP) study showed that in patients with elevated transcranial Doppler velocities, prophylactic blood transfusion to maintain HbS less than 30% could reduce the risk of stroke to less than 1% [42]. Exchange transfusion is recommended in SCD patients with an image-confirmed acute CVA [25, 43]. Additionally, myeloablative bone marrow transplant appears to offer a significant benefit in limiting cerebral vasculopathy and, in turn, risk for a CVA [13]. Tissue plasminogen activator (tPA) has not been studied specifically in patients with SCD. Therefore, the determination of whether to treat SCD patients presenting with a CVA with tPA should be made in consultation with a neurologist and hematologist [4].

Pulmonary Embolism

Patients with SCD should intuitively seem susceptible to venous thromboembolism (VTE) given their increased coagulability, endothelial dysfunction, and impaired blood flow (Virchow's triad) [44, 45]. Indeed, the incidence of pulmonary embolism (PE) is higher in patients with SCD. There is a 50–100-fold increase in annual incidence in inpatients with SCD compared to those without SCD [44, 46, 47]. VTE is present in 50% of autopsies of patients with SCD, and yet

only 5% of these are detected clinically [46]. Additionally, patients with AChS have a reported PE prevalence rate approaching 17%, and PE may be a contributing factor to the pulmonary hypertension seen in patients with SCD [36]. Despite this, inpatients with SCD undergo fewer chest computed tomography (CT) angiography tests for PE compared to those without SCD [46].

Unfortunately, several clinical prediction rules for VTE are unreliable in the setting of SCD since symptoms of AChS and VOC may overlap with those of VTE. D-dimer testing is also of limited utility since elevated D-dimer levels are found in the vast majority (92%) of patients with VOC [44, 45].

Therefore, a high index of suspicion for VTE must be maintained when evaluating patients with SCD. Computed tomography pulmonary angiography (CTPA) and ventilation-perfusion scanning are both reasonable options for the further workup of pulmonary embolism [48, 49]. Despite concerns and reports regarding intravenous contrast inducing nephropathy and/or increasing sickling of erythrocytes, there is data indicating no difference in adverse events compared to patients without SCD. Lower extremity venous duplex evaluation should also be considered to reduce radiation risk and the risk of other complications in patients with VOC.

Acute Coronary Syndrome

The literature indicates that patients with SCD have minimal atherosclerosis of the large coronary arteries [50–53]. This is despite case reports and autopsy series which reveal findings of myocardial infarction presumably from vaso-occlusive effects on coronary microcirculation [53–55]. Therefore, acute coronary syndrome (ACS) should be considered in the differential for patients with SCD presenting with cardiac symptoms regardless of age (since VOC at the microvascular level is not age dependent).

Since SCD by itself does not result in a baseline increase in cardiac biomarkers (troponin-I, troponin-T, and creatinine kinase isoenzyme MB) [53, 56], an appropriate strategy would be to evaluate patients that are deemed at risk for ACS with serial electrocardiograms and serial cardiac biomarkers. Stress testing or cardiac CT angiography (CCTA) should be considered based on national guidelines.

The treatment of acute ST-elevation myocardial infarction should err on the side of interventional treatment since thrombolytics have not been studied specifically in patients with SCD. However, when interventional options are unavailable in a timely manner, thrombolytics should be considered in consultation with a hematologist and cardiologist when possible.

Renal Complications

Renal complications are extremely common in SCD with 30% of adults developing chronic renal failure [13]. Prerenal, post-renal, and intrinsic renal factors all contribute to this dysfunction. Acute kidney injury can occur during an acute VOC due to low partial pressure of oxygen, low pH, and high osmolality in the renal medulla which all contribute to erythrocyte dehydration and vaso-occlusion [12, 25]. Diffuse vaso-occlusion can result in multisystem organ failure including renal complications.

Microalbuminuria and proteinuria are common diagnostic findings [14, 25]. It is important to avoid nephrotoxic drugs and monitor renal function on a regular basis. ACE inhibitors are used prophylactically (including in children) to reduce the progression of disease, while transplantation has significant benefits in survival over dialysis [12].

Ocular Complications

Proliferative retinopathy is the most common ophthalmologic complication of SCD (more common in HbSC – as high as 70%) and results from occlusion of the peripheral retinal vasculature [13, 14]. Permanent loss of vision can occur from ocular complications not only due to progression of this proliferative sickle retinopathy but also from complications of trauma, infection, and VOC.

Hyphema can lead to significant complications in patients with SCD and also with sickle cell trait since sickling is more likely to occur in the hypoxic and acidotic anterior chamber with resultant outflow tract obstruction. The elevated intraocular pressure (IOP) that results can cause central retinal artery occlusion and infarction of the optic nerve [57, 58]. Presence of a significant hyphema warrants ophthalmologic consultation with possible surgical drainage [59, 60].

Orbital infarction, although rare, can also occur in SCD patients experiencing a VOC, and compression of the optic nerve as well as nerves affecting extraocular movements are also possible [61, 62]. Surgical intervention is indicated if symptoms progress rapidly.

Splenic Sequestration

Splenic sequestration is a potentially catastrophic complication of SCD and is characterized by an acute drop in hemoglobin that can cause circulatory collapse. It is more common in the pediatric population, especially with HbSS, since they have not yet auto-infarcted their spleen [14, 41]. The mechanism for this entity involves sickled erythrocytes becoming trapped within the spleen and thereby being removed from the circulatory system, manifested in laboratory testing as a profound anemia.

EPs must consider this entity when presented with an adult or child with acute splenic enlargement and circulatory collapse. Splenic sequestration can be differentiated from other causes of anemia associated with SCD (hemolysis, red cell aplasia) based on a combination of laboratory tests: severe anemia, elevated reticulocyte count, and findings of hemolysis (elevated indirect bilirubin, ALT, and LDH) [41]. Guidance regarding the management of SCD patients with splenic sequestration is limited, but transfusion (without over-transfusion) and splenectomy are being advised, in addition to observation based on the patient's clinical status [25]. Eventually a splenectomy is critical since splenic sequestration recurs at a rate of nearly 50% [14].

Priapism

Priapism associated with SCD is typically of the low-flow type associated with stasis, hypoxia, and ischemia, with about 35% of boys/men affected [14, 63]. Conservative management with fluid hydration and analgesia are an appropriate first step, with medical management including injection of phenylephrine into the corpora cavernosa, subcutaneous terbutaline, and/or oral pseudoephedrine. Interventional approaches such as aspiration and irrigation with alphaadrenergic agents are an appropriate next step. If these therapies are unsuccessful, then emergent urologic consultation for possible shunting is recommended.

Hepatobiliary

Biliary sludge and cholelithiasis/choledocholithiasis are common in SCD patients due to chronic hemolysis and increased bilirubin turnover [64]. Gallstones are noted on ultrasound in up to 70% of patients [25]. Asymptomatic patients should be managed conservatively, while surgical cholecystectomy, including laparoscopic approaches, has good outcomes [65].

As with splenic sequestration, hepatic crisis and hepatic sequestration can occur and require aggressive management with hydration, rest, and close observation [25]. Exchange transfusion has demonstrated positive results [66]. Of note, frequent transfusions can also result in hemosiderosis or hemochromatosis.

Avascular Necrosis (AVN)

Sickling results in occlusion of capillaries which in turn causes inadequate blood supply with consequential bone death. The femoral heads are common sites of this osteonecrosis. About 50% of patients with HbSS demonstrate some avascular necrosis by age 33 [67]. Pain control and physical therapy are the mainstays of treatment, with surgical treatment for symptomatic patients with advanced AVN [25].

Aplastic Crises

Commonly seen in children, aplastic crisis presents with gradual onset fatigue, shortness of breath, and fever. Patients can present with tachycardia and even florid heart failure. Parvovirus B19 has been identified as the inciting infection [68], causing an interruption of erythropoiesis via destruction of erythrocyte precursors and resulting in severe anemia and cardiovascular decompensation. This self-limited infection typically lasts 7–10 days and results in lifelong immunity [14].

Controversies in the Management of Sickle Cell Disease

Acute Pain

Unfortunately, stigma and judgment often influence the diagnosis and management of SCD patients who present to the ED in pain. It is important to remember that these patients can experience similar painful disease processes as patients without SCD and failure to recognize non-SCD-related conditions can result in poor outcomes. As mentioned previously, patients are generally able to ascertain if the acute pain they are feeling is typical of prior VOC episodes. Additional testing should be considered when patients complain that a painful episode is different in character (not just severity) from typical episodes. In some cases, VOC may be triggered by an unrelated diagnosis, but the acute pain from the VOC may mask this underlying condition. Therefore, it is important that EPs proactively set aside their biases and approach patients with SCD in an open-minded and non-judgmental manner.

Physicians treating patients with SCD are at specific risk for cognitive biases, also known as cognitive dispositions to respond. Examples of cognitive bias include being overcommitted to a given diagnosis, failure to consider alternative diagnoses, continuing with a diagnosis reached by others, being overly influenced by past patient experiences, ending the diagnostic workup prematurely, and being influenced by context or patient characteristics [69, 70].

Opioid Crisis and Implications for the Care of Patients with SCD

Over the past 10 years, the opioid crisis in the United States has received increasing attention. Given that that from 1999 to 2018 over 400,000 people died from an opioid overdose, that 2 out of 3 drug overdose deaths in 2018 involved an opioid, that nearly 47,000 people died from opioid overdoses in 2018, and that 32% of those deaths involved prescription opioids [71, 72], there is immense pressure on emergency physicians to curtail the use and prescription of opioids. However, the challenges related to the objective assessment of pain as well as pressures related to patient satisfaction scores and the emphasis on early pain management by regulatory agencies have placed emergency physicians in a quandary regarding the appropriate management of SCD patients presenting with acute pain.

However, prudent strategies can guide emergency departments and EPs in this regard. First, it is helpful to develop departmental and/or institutional evidence-based guidelines in a collaborative manner with a variety of personnel involved in the care of patients with SCD. Second, these guidelines should recognize the need to differentiate opioid-naive from opioid-tolerant patients while also acknowledging that the latter group is not simply "addicted" to opioids but that there is a physiologic basis for the dependence that has developed over time. Finally, prudent limitations such as avoidance of intramuscular administration and minimizing use of intravenous hydromorphone and diphenhydramine while still utilizing opioids in the management can guide EPs in appropriately caring for patients with VOC.

Additionally, high utilizers of emergency department services should be managed proactively with a diverse care team and individualized care plans to ensure that the patient's needs are met while being judicious in the use of opioid medications. Increasing interoperability of electronic medical records and prescription drug monitoring programs are helpful in assessing a patient's use of healthcare resources and potential overuse of opioids. Goals driven by care plans and collaborations between EPs and outpatient physicians are shown in Table 41.7 [73].

Intravenous Fluids

Given that erythrocyte dehydration plays a significant role in the pathophysiology of SCD, hydration has long been a com-

 Table 41.7 Goals in promoting care coordination and continuity

Patient received coordinated, comprehensive care

Patient keeps regular appointments (for patients with three or more ED visits per year, it is recommended they have at least four scheduled outpatient appointments per year, with a goal of not missing more than one appointment except when missed because of hospitalization)

Patient is compliant with health promotion and disease management programs

Patient abstains from using illicit opioids

Patient obtains opioid prescriptions from a single practice

From Lovett et al. [73], with permission Elsevier

ponent of therapy. However, while isolated and suboptimal studies have shown a potential benefit to intravenous hydration during VOC and for other complications, there is inadequate data to assess the safety of aggressive hydration [74]. Additionally, there are reports linking excessive fluid administration to the development of atelectasis [75], and this in turn to AChS. Thus, it is recommended that boluses of intravenous fluids be reserved for suspected or proven hypovolemia or hypertonicity [4].

Blood Transfusion

Erythrocyte transfusions, with the goal of reducing HbS to less than 30%, are indicated in the urgent management of acute anemia, AChS, acute neurological deficits, multi-organ failure, and perioperative management [12, 25, 76]. However, care must be taken to avoid volume overload and increased blood viscosity associated with transfusion. Transfusion is not recommended in patients with asymptomatic anemia and those with acute kidney injury in the absence of multisystem organ failure [25].

Transfusions may be considered in patients hospitalized with complications of SCD (e.g., AChS, cerebral infarction, ACS) but not simple VOC. Anemia should not be overcorrected with a goal of no more than 10 g/dL with care be taken to use leukocyte-reduced blood that is matched for all relevant antigens (Rh, C, E, and Kell) [25, 33]. Regular, long-term transfusions are beneficial in the prevention of strokes in pediatric patients, but carry risks of alloimmunization and iron overload, and hence should be coordinated with the patient's hematologist [20, 25]. Figure 41.3 represents



Fig. 41.3 Indications and goals for blood transfusion. (From Ware et al. [15], with permission The Lancet and Elsevier)

current recommendations regarding transfusion therapy for acute complications of SCD [15].

Hydroxyurea (Hydroxycarbamide)

Hydroxyurea is a ribonucleotide reductase inhibitor that can increase HbF levels, in turn inhibiting polymerization of deoxygenated HbS [77–79]. Additionally, it triggers increased donation of NO and a reduction in erythrocyte adhesiveness and leukocyte count [13, 25, 79]. Results of the Multicenter Study of Hydroxyurea in Sickle Cell Anemia (MSH) were promising with a 50% reduction in painful episodes, reduced need for blood transfusions, and lower mortality rate at 9 years [80]. While a potentially effective therapy, it requires time to trigger an increase in HbF and should be given per a standardized protocol – it is therefore not a realistic option for acute ED management, but should be strongly considered if the patient is being cared for in an ED observation unit [25].

Looking Ahead

It is striking to note the dearth of high-quality evidence, especially randomized controlled trials (RCTs), related to the management of SCD. Sickle cell disease exemplifies the potential racial and ethnic disparities that exist in healthcare spending and research, as well as stigmas associated with patients afflicted with it. The most tangible hope for a cure exists with hematopoietic stem cell transplantation (HSCT) but is feasible in only a small proportion of patients with SCD [81].

Several other therapies are under early investigation (phase 1 and 2 trials) [15], and gene therapy has shown promise; however, introduction to clinical practice remains years away [82].

Summary

Sickle cell disease is a genetic disorder commonly encountered by ED providers. Acute painful episodes are by far the most common reason for ED visits. Early and aggressive pain management is a key priority in these visits; however, EPs must also actively seek to diagnose other emergent diagnoses in patients with SCD, including AChS and infection, and differentiate them from VOC. EPs should be especially aware of cognitive biases that may misdirect the diagnostic process.

Bolus administration of intravenous fluids should be considered only for hypovolemic patients. Blood transfusion may be considered to reduce the percentage of HbS, but close attention must be given to its associated risks – impact on volume status and viscosity. Conditions such as acute coronary syndrome, pulmonary embolism, and cerebrovascular accident must be considered in the evaluation of patients with SCD, and while management is guided by the standard of care for the individual conditions, early consultation with hematology is imperative.

Finally, we cannot over-emphasize the importance of coordinating care with hematology and the outpatient team in the effective emergency and long-term management of patients with sickle cell disease.

References

- National Human Genome Research Institute: About Sickle Cell Disease. https://www.genome.gov/Genetic-Disorders/Sickle-Cell-Disease. Accessed 21 Apr 2020.
- Centers for Disease Control and Prevention: Data & Statistics on Sickle Cell Disease. https://www.cdc.gov/ncbddd/sicklecell/data. html. Accessed 21 Apr 2020.
- Ballas SK, Gupta K, Adams-Graves P. Sickle cell pain: a critical reappraisal. Blood. 2012;120(18):3647–56.
- Glassberg J. Evidence-based management of sickle cell disease in the emergency department. Emerg Med Pract. 2011;13(8):1–20.
- Feliu MH, Wellington C, Crawford RD, Wood M, Edwards L, Byrd G, Edwards CL. Opioid management and dependency among adult patients with sickle cell disease. Hemoglobin. 2011;35(5–6):485–94.
- Field JJ, Knight-Perry JE, Debaun MR. Acute pain in children and adults with sickle cell disease: management in the absence of evidence-based guidelines. Curr Opin Hematol. 2009;16(3):173–8.
- Herrick JB. Peculiar elongated and sickle-shaped red blood corpuscles in a case of severe anemia. 1910. Yale J Biol Med. 2001;74(3):179–84.
- Neel JV. The inheritance of sickle cell anemia. Science. 1949;110(2846):64–6.
- Ingram VM. A specific chemical difference between the globins of normal human and sickle-cell anaemia haemoglobin. Nature. 1956;178(4537):792–4.
- Hsia CC. Respiratory function of hemoglobin. N Engl J Med. 1998;338(4):239–47.
- Steinberg MH. Overview of sickle cell anemia pathophysiology. In: Costa FF, Conran N, editors. Sickle cell anemia: from basic science to clinical practice. Cham: Springer; 2016. p. 49–73. https://doi. org/10.1007/978-3-319-06713-1.
- Rees DC, Williams TN, Gladwin MT. Sickle-cell disease. Lancet. 2010;376(9757):2018–31. https://doi.org/10.1016/ S0140-6736(10)61029-X.
- Kassim AA, DeBaun MR. Sickle cell disease, vasculopathy, and therapeutics. Annu Rev Med. 2013;64:451–66.
- Stuart MJ, Nagel RL. Sickle-cell disease. Lancet. 2004;364(9442):1343–60.
- Ware RE, de Montalembert M, Tshilolo L, Abboud MR. Sickle cell disease. Lancet. 2017;390(10091):311–23. https://doi.org/10.1016/ S0140-6736(17)30193-9.
- Allison AC. Protection afforded by sickle-cell trait against subtertian malarial infection. Br Med J. 1954;1(4857):290–4.
- 17. Peil FB, Williams TN. Sickle cell anemia: history and epidemiology. In: Costa FF, Conran N, editors. Sickle cell anemia: from

basic science to clinical practice. Cham: Springer; 2016. p. 23–47. https://doi.org/10.1007/978-3-319-06713-1.

- Serjeant GR. Geography and the clinical picture of sickle cell disease. An overview. Ann NY Acad Sci. 1989;565:109–19.
- Hassell KL. Population estimates of sickle cell disease in the U.S. Am J Prev Med. 2010;38(4 Suppl):S512–21.
- Quinn CT, Rogers ZR, McCavit TL, Buchanan GR. Improved survival of children and adolescents with sickle cell disease. Blood. 2010;115(17):3447–52.
- Costa FF, Fertrin KY. Clinical manifestations and treatment of adult sickle cell disease. In: Costa FF, Conran N, editors. Sickle cell anemia: from basic science to clinical practice. Cham: Springer; 2016. p. 285–318. https://doi.org/10.1007/978-3-319-06713-1.
- Ballas SK. Pathophysiology and principles of management of the many faces of the acute vaso-occlusive crisis in patients with sickle cell disease. Eur J Haematol. 2015;95(2):113–23. https://doi. org/10.1111/ejh.12460.
- Dampier C, Palermo TM, Darbari DS, Hassell K, Smith W, Zempsky W. AAPT Diagnostic criteria for chronic sickle cell disease pain. J Pain. 2017;18(5):490–8. https://doi.org/10.1016/j. jpain.2016.12.016.
- Smith WR, Penberthy LT, Bovbjerg VE, McClish DK, Roberts JD, Dahman B, et al. Daily assessment of pain in adults with sickle cell disease. Ann Intern Med. 2008;148(2):94–101.
- 25. National Heart, Lung, and Blood Institute Expert Panel Report. Evidence-based management of sickle cell disease. 2014. https:// www.nhlbi.nih.gov/sites/default/files/media/docs/sickle-celldisease-report%20020816_0.pdf. Accessed 17 May 2020.
- 26. Fiocchi J, Urits I, Orhurhu V, Orhurhu MS, Giacomazzi S, Hoyt B, et al. A comprehensive review of the treatment and management of pain in sickle cell disease. Curr Pain Headache Rep. 2020;24(5):17.
- 27. Abbott J. Even "frequent flyers" die. Ann Emerg Med. 2009;54(6):840.
- Ballas SK. Meperidine for acute sickle cell pain in the emergency department: revisited controversy. Ann Emerg Med. 2008;51(2):217.
- Phillips WJ, Jones J, McCarter J. Meperidine for acute sickle cell crisis: what is the rationale? Ann Emerg Med. 2008;52(6):766–7.
- Morgan MT. Use of meperidine as the analgesic of choice in treating pain from acute painful sickle cell crisis. Ann Emerg Med. 2008;51(2):202–3.
- Ballas SK. Pain management of sickle cell disease. Hematol Oncol Clin North Am. 2005;19(5):785–802, v.
- 32. Charache S, Scott JC, Charache P. "Acute chest syndrome" in adults with sickle cell anemia. Microbiology, treatment, and prevention. Arch Intern Med. 1979;139(1):67–9.
- Stuart MJ, Setty BN. Acute chest syndrome of sickle cell disease: new light on an old problem. Curr Opin Hematol. 2001;8(2):111– 22. https://doi.org/10.1097/00062752-200103000-00009.
- 34. Stuart MJ, Setty BN. Sickle cell acute chest syndrome: pathogenesis and rationale for treatment. Blood. 1999;94(5):1555–60.
- 35. Vichinsky EP, Neumayr LD, Earles AN, Williams R, Lennette ET, Dean D, et al. Causes and outcomes of the acute chest syndrome in sickle cell disease. National Acute Chest Syndrome Study Group. N Engl J Med. 2000;342(25):1855–65.
- Mekontso Dessap A, Deux JF, Abidi N, Lavenu-Bombled C, Melica G, Renaud B, et al. Pulmonary artery thrombosis during acute chest syndrome in sickle cell disease. Am J Respir Crit Care Med. 2011;184(9):1022–9.
- Bernini JC, Rogers ZR, Sandler ES, Reisch JS, Quinn CT, Buchanan GR. Beneficial effect of intravenous dexamethasone in children with mild to moderately severe acute chest syndrome complicating sickle cell disease. Blood. 1998;92(9):3082–9.
- Sobota A, Graham DA, Heeney MM, Neufeld EJ. Corticosteroids for acute chest syndrome in children with sickle cell disease: varia-

tion in use and association with length of stay and readmission. Am J Hematol. 2010;85(1):24–8.

- King D, Conway EE Jr. Acute chest syndrome (ACS) in pediatric patients admitted to a pediatric critical care unit (PCCU) (abstract). Chest. 1996;110(Suppl 4):1485.
- Liem RI, O'Gorman MR, Brown DL. Effect of red cell exchange transfusion on plasma levels of inflammatory mediators in sickle cell patients with acute chest syndrome. Am J Hematol. 2004;76(1):19–25.
- Subramanium S, Chao JH. Managing acute complications of sickle cell disease in pediatric patients. Emerg Med Pract. 2016;13(11):1–28.
- 42. Adams RJ, McKie VC, Hsu L, Files B, Vichinsky E, Pegelow C, et al. Prevention of a first stroke by transfusions in children with sickle cell anemia and abnormal results on transcranial Doppler ultrasonography. N Engl J Med. 1998;339(1):5–11.
- 43. Ware RE, Helms RW. Stroke with transfusions changing to hydroxyurea (SWiTCH). Blood. 2012;119(17):3925–32.
- 44. Naik RP, Streiff MB, Lanzkron S. Sickle cell disease and venous thromboembolism: what the anticoagulation expert needs to know. J Thromb Thrombolysis. 2013;35(3):352–8.
- Lim MY, Ataga KI, Key NS. Hemostatic abnormalities in sickle cell disease. Curr Opin Hematol. 2013;20(5):472–7.
- Novelli EM, Huynh C, Gladwin MT, Moore CG, Ragni MV. Pulmonary embolism in sickle cell disease: a case-control study. J Thromb Haemost. 2012;10(5):760–6.
- 47. Brunson A, Lei A, Rosenberg AS, White RH, Keegan T, Wun T. Increased incidence of VTE in sickle cell disease patients: risk factors, recurrence and impact on mortality. Br J Haematol. 2017;178(2):319–26.
- Tivnan P, Billett HH, Freeman LM, Haramati LB. Imaging for pulmonary embolism in sickle cell disease: a 17-year experience. J Nucl Med. 2018;59(8):1255–9.
- 49. Swift CA, Nance JW, Collins H, Ravenel JG. Incidence of pulmonary embolism in sickle cell anemia patients undergoing computed tomography pulmonary angiography in the emergency department. J Thorac Imaging. 2019;34(5):W125–6.
- Sachde V, Rosing DR, Thein SL. Cardiovascular complications of sickle cell disease. Trends Cardiovasc Med. 2020; pii: S1050–1738(20)30020–7.
- Barrett O Jr, Saunders DE Jr, McFarland DE, Humphries JO. Myocardial infarction in sickle cell anemia. Am J Hematol. 1984;16(2):139–47.
- 52. Manci EA, Culberson DE, Yang YM, Gardner TM, Powell R, Haynes J Jr, et al.; Investigators of the Cooperative Study of Sickle Cell Disease. Causes of death in sickle cell disease: an autopsy study. Br J Haematol, 2003;123(2):359–65.
- Aslam AK, Rodriguez C, Aslam AF, Vasavada BC, Khan IA, Cardiac troponin I. in sickle cell crisis. Int J Cardiol. 2009;133(1):138–9.
- Pavlů J, Ahmed RE, O'Regan DP, Partridge J, Lefroy DC, Layton DM. Myocardial infarction in sickle-cell disease. Lancet. 2007;369(9557):246.
- 55. Sherman SC, Sule HP. Acute myocardial infarction in a young man with sickle cell disease. J Emerg Med. 2004;27(1):31–5.
- Lippi G, De Franceschi L, Salvagno GL, Pavan C, Montagnana M, Guidi GC. Cardiac troponin T during sickle cell crisis. Int J Cardiol. 2009;136(3):357–8.
- 57. Goldberg MF. The diagnosis and treatment of secondary glaucoma after hyphema in sickle cell patients. Am J Ophthalmol. 1979;87(1):43–9.
- Lai JC, Fekrat S, Barron Y, Goldberg MF. Traumatic hyphema in children: risk factors for complications. Arch Ophthalmol. 2001;119(1):64–70.
- Walton W, Von Hagen S, Grigorian R, Zarbin M. Management of traumatic hyphema. Surv Ophthalmol. 2002;47(4):297–334.

- Deutsch TA, Weinreb RN, Goldberg MF. Indications for surgical management of hyphema in patients with sickle cell trait. Arch Ophthalmol. 1984;102(4):566–9.
- Curran EL, Fleming JC, Rice K, Wang WC. Orbital compression syndrome in sickle cell disease. Ophthalmology. 1997;104(10):1610–5.
- Sokol JA, Baron E, Lantos G, Kazim M. Orbital compression syndrome in sickle cell disease. Ophthal Plast Reconstr Surg. 2008;24(3):181–4.
- 63. Olujohungbe AB, Adeyoju A, Yardumian A, Akinyanju O, Morris J, Westerdale N, et al. A prospective diary study of stuttering priapism in adolescents and young men with sickle cell anemia: report of an international randomized control trial--the priapism in sickle cell study. J Androl. 2011;32(4):375–82.
- Sarnaik S, Slovis TL, Corbett DP, Emami A, Whitten CF. Incidence of cholelithiasis in sickle cell anemia using the ultrasonic grayscale technique. J Pediatr. 1980;96(6):1005–8.
- Al-Mulhim AS, Al-Mulhim FM, Al-Suwaiygh AA. The role of laparoscopic cholecystectomy in the management of acute cholecystitis in patients with sickle cell disease. Am J Surg. 2002;183(6):668–72.
- Ahn H, Li CS, Wang W. Sickle cell hepatopathy: clinical presentation, treatment, and outcome in pediatric and adult patients. Pediatr Blood Cancer. 2005;45(2):184–90.
- 67. Milner PF, Kraus AP, Sebes JI, Sleeper LA, Dukes KA, Embury SH, et al. Sickle cell disease as a cause of osteonecrosis of the femoral head. N Engl J Med. 1991;325(21):1476–81.
- Serjeant GR, Topley JM, Mason K, Serjeant BE, Pattison JR, Jones SE, et al. Outbreak of aplastic crises in sickle cell anaemia associated with parvovirus-like agent. Lancet. 1981;2(8247):595–7.
- 69. Croskerry P, Singhal G, Mamede S. Cognitive debiasing 1: origins of bias and theory of debiasing. BMJ Qual Saf. 2013;22(Suppl 2):ii58–64.
- Reilly JB, Ogdie AR, Feldt JMV, Myers JS. Teaching about how doctors think: a longitudinal curriculum in cognitive bias and diagnostic error for residents. BMJ Qual Saf. 2013;22(12):1044–50.
- Centers for Disease Control and Prevention: Opioid overdose: Opioid basics – understanding the epidemic. https://www.cdc.gov/ drugoverdose/epidemic/index.html. Accessed 14 May 2020.

- Centers for Disease Control and Prevention: Opioid overdose: data overview. https://www.cdc.gov/drugoverdose/data/. Accessed 14 May 2020.
- Lovett PB, Sule HP, Lopez BL. Sickle cell disease in the emergency department. Emerg Med Clin North Am. 2014;32(3):629–47. https://doi.org/10.1016/j.hoc.2017.08.009.
- Okomo U, Meremikwu MM. Fluid replacement therapy for acute episodes of pain in people with sickle cell disease. Cochrane Database Syst Rev. 2017;7(7):CD005406.
- Vichinsky EP, Styles LA, Colangelo LH, Wright EC, Castro O, Nickerson B. Acute chest syndrome in sickle cell disease: clinical presentation and course. Cooperative Study of Sickle Cell Disease. Blood. 1997;89(5):1787–92.
- Howard J, Malfroy M, Llewelyn C, Choo L, Hodge R, Johnson T, et al. The Transfusion Alternatives Preoperatively in Sickle Cell Disease (TAPS) study: a randomised, controlled, multicentre clinical trial. Lancet. 2013;381(9870):930–8.
- Platt OS, Orkin SH, Dover G, Beardsley GP, Miller B, Nathan DG. Hydroxyurea enhances fetal hemoglobin production in sickle cell anemia. J Clin Invest. 1984;74(2):652–6.
- Rodgers GP, Dover GJ, Noguchi CT, Schechter AN, Nienhuis AW. Hematologic responses of patients with sickle cell disease to treatment with hydroxyurea. N Engl J Med. 1990;322(15):1037–45.
- Charache S, Dover GJ, Moore RD, Eckert S, Ballas SK, Koshy M, et al. Hydroxyurea: effects on hemoglobin F production in patients with sickle cell anemia. Blood. 1992;79(10):2555–65.
- Charache S, Terrin ML, Moore RD, Dover GJ, Barton FB, Eckert SV, et al. Effect of hydroxyurea on the frequency of painful crises in sickle cell anemia. Investigators of the Multicenter Study of Hydroxyurea in Sickle Cell Anemia. N Engl J Med. 1995;332(20):1317–22.
- Freed J, Talano J, Small T, Ricci A, Cairo MS. Allogeneic cellular and autologous stem cell therapy for sickle cell disease: 'whom, when and how'. Bone Marrow Transplant. 2012;47(12):1489–98.
- Demirci S, Uchida N, Tisdale JF. Gene therapy for sickle cell disease: an update. Cytotherapy. 2018;20(7):899–s.

Radiological and Nuclear Terrorism

Carol J. Iddins, Jason E. Davis, Ronald E. Goans, and Cullen Case Jr.

Introduction

Recent world events and political tensions are again bringing attention to the possibility of a nuclear incident, including a detonation. Radiological materials are ubiquitous and are found in most cities throughout the country at academic institutions, hospitals, and many businesses and in industrial settings. Nuclear materials are more difficult to access because of their scarce nature and the security of the settings in which they are found. A significant radiological or nuclear (R/N) incident in the Unites States will require activation of local, state, territory, regional, and national government assets. Initially, ad hoc triage and treatment areas will rise up in the moderately to lightly damaged zones. Patients will undergo mass casualty triage and stabilization before medical evacuation to subsequent higher levels of care. This will require subject matter expert advice and response as the size and circumstances of the incident unfold. Initial mass casualty triage should be done to sort the populations who will require minor interventions, admission to the hospital, and possibly eventual intensive care and those who may have an extremely dire prognosis and may require comfort or hospice care. The hematologist-oncologist will be among those with the expertise to effectively treat those patients who will experience bone marrow aplasia as part of radiation exposure that is higher than 1-2 Gray (Gy). Acute radiation syndrome

J. E. Davis

(ARS) is the complex of organ system injuries that occur from the radiation exposure. The organ systems show a classical clinical prodrome that correlates to radiation exposure dose. This syndrome will affect bone marrow, gastrointestinal, cutaneous, pulmonary, and neurovascular systems. These patients will require the expertise of intensivists and those experienced in managing systemic inflammatory response syndrome (SIRS), multiple organ dysfunction (MOD), and multiple organ failure (MOF) that may result. This chapter provides an overview for the hematologistoncologist in managing patients who may present after an R/N incident.

Some understanding of the mechanisms of damage incurred from radiation is helpful in managing these patients. This is not routinely included in our medical or graduate medical education. Many oncologists and radiation oncologists deal with radiotherapy adverse effects, but, otherwise, most physicians never manage a patient from a radiological incident. A review of some important terms and topics is included for better understanding of the disease processes.

Radiological materials have many uses in academia, medicine, and industry. They may be used in the initiation or attempted initiation of an energetic chain reaction of fissionable materials. Small quantities of fissionable nuclear materials generally pose more of a chemical hazard than a radiological hazard. If sufficient quantities of fissionable material are present in the correct geometry, a chain reaction is initiated resulting in the release of immense amounts of energy and the production of highly radioactive isotopes. Nuclear fission supplies the destructive potential of improvised nuclear devices (IND) (a terrorist nuclear bomb) and more sophisticated nuclear weapons. Fission is also the process used to generate heat for the production of electricity in nuclear power plants (NPPs). In fission reactors, steam generated from the nuclear process is used to turn a turbine which, in turn, rotates a generator [1].

Measurement of Radioactivity Radioactivity, the activity of a radiation source, is the term used to describe how much



C. J. Iddins

Radiation Emergency Assistance Center/Training Site, Oak Ridge Institute for Science and Education, Oak Ridge, TN, USA

Oak Ridge Associated Universities, Radiation Emergency Assistance Center/Training Site, Oak Ridge Institute for Science and Education, Oak Ridge, TN, USA

R. E. Goans MJW Corporation, Clinton, TN, USA

C. Case Jr. (\boxtimes) Radiation Injury Treatment Network, National Marrow Donor Program, Minneapolis, MN, USA e-mail: ccase@nmdp.org
energy is being released by radioactive material in a given time. Radioactivity or activity is measured in curies (Ci) in the English measurement system or the becquerel (Bq) in the SI system (SI=International System of Units or Système International). A Bq is equivalent to one disintegration of an atomic nucleus per second. A Ci is equivalent to 3.7×10^{10} disintegrations per second (dps) and is based on the decay rate of radium-226. A Bq is so small that it is much more common to see units in multiples of Bq such as megabecquerels (MBq), gigabecquerels (GBq), etc. Likewise, a curie is so large that it is much more common to see units in fractions of Ci such as millicuries (mCi), microcuries (µCi), etc.

Units of Dose Measurement The unit rad is often used in the English system to describe the amount of ionizing radiation that is absorbed in a cell, tissue, organ, or the body. It is equivalent to 100 ergs of energy deposited in 1 g of tissue. The gray (Gy) is equivalent to 1 J of energy deposited in 1 kg of tissue. One Gy is equivalent to 100 rad. The rem (roentgen equivalent man) is a unit of equivalent dose which is used to measure the long-term biological risk related to ionizing radiation exposure (in the USA). The sievert (Sv) is the international unit (SI) for equivalent dose. One Sv is equivalent to 100 rem. The terms Gy and Sv will be used.

Radiological and Nuclear Scenarios of Concern

Key to understanding radiological and nuclear incidents are the types of injuries and illnesses that they can cause. There are many occupational and medical exposures occurring with more frequency. However, the following radiation scenarios are of concern for emergency care responders:

- Radiological exposure device (RED)
- Radiological dispersal device (RDD)
- Improvised nuclear device (IND)
- Nuclear weapon detonation (NWD)
- Nuclear power plant (NPP) incident

Radiological Exposure Device (RED)

A RED is any radioactive material or an object containing radioactive material that can expose people to radiation without their knowledge. These materials may be inadvertently left in an accessible area or may be intentionally placed in a public space with the intent to cause harm to one or more people. The radiation-induced injuries and illnesses that may result from exposure or touching the source vary, depending upon the nature of the source, the radiation emitted, and the energy of the radiation emitted. It is possible for an RED to cause severe damage up to, and including, ARS, as well as cutaneous radiation injuries (CRI) with damage to the skin and deeper tissues and organs.

Radiological Dispersal Device (RDD)

A radiological dispersal device (RDD) is any device that spreads radioactive material in the environment with the intent to cause panic, destruction, damage, or injury. A "dirty bomb" is one type of RDD, in which explosives are used as the mechanism to disperse the radioactive material. Radioactive materials needed to produce an RDD can be obtained from industrial, commercial, medical, and research applications. An explosive RDD may result in the immediate threat to human life. Hazards will occur from fire, smoke, shock (physical, or thermal), and debris from the explosion. The long-term consequences of an explosive RDD could involve increased risk of cancer to exposed individuals resulting from direct exposure to radioactive materials, the inhalation of radioactive materials, and potential contact with other hazardous materials present at the scene. In most plausible scenarios, the radioactive material would not result in acutely harmful radiation doses.

Improvised Nuclear Device (IND) and Nuclear Weapon Detonation (NWD)

An improvised nuclear device (IND) can be described as a homemade atomic bomb in which the most likely configuration includes two subcritical masses of a fissile material that are quickly and forcefully brought together. Bringing these masses together allows for an uncontrolled fission reaction to occur, resulting in a massive release of energy in the form of thermal radiation, ionizing radiation, and a forceful pressure wave. Improvised nuclear devices are likely to cause more nuclear fallout because of incomplete fission due to a number of engineering complexities. They are also likely to be exploded at ground level rather than at an altitude, which causes ground materials to be irradiated and drawn up into the sky by the lifting of heated gases. Mistakes in the design and timing of the device may also result in a fizzle, where the criticality is too short-lived to consume all of the fuel or the device fails to achieve criticality, resulting in the fuel being dispersed without being consumed. The primary distinction between an IND and a traditional nuclear weapon is the sophistication of the manufactured device.

The nuclear weapons detonated over Japan during World War II were on the order of 10–15 kilotons (KT) of TNT and involved the use of only a few pounds of U-235 or Pu-238. Weapons developed later during the Cold War were on the order of megatons. When discussing nuclear weapons, it is important to realize that they are designed and manufactured

with very sophisticated nuclear, chemical, and electrical engineering skills and techniques. The detonation of a nuclear weapon would be devastating, with massive infrastructure damage and massive numbers of casualties. The fallout of radioactive material descending to the ground after being blown into the atmosphere by the detonation can result in significant radiation exposures. The acute and subacute consequences of such detonations will involve physical trauma and thermal burns, as well as radiation-induced illnesses/injuries.

Nuclear Power Plant (NPP) Incidents

NPP incidents are rare and historically have produced variable outcomes. Not all of these outcomes manifest physically. Sometimes the psychological or social impact can be equally devastating, and health effects may be more related to lack of daily necessities rather than radiation. The Chernobyl (1986) and Fukushima (2011) incidents resulted in multiple human health effects as a result of the incidents. Chernobyl resulted in 28 acute deaths from ARS, primarily related to exposures to, and contamination with, radioactive materials [2]. At Fukushima, the public health emergency from the Great East Japan Earthquake and tsunami resulted in many health effects and deaths; however, none of these were related to acute radiation effects. The Three Mile Island incident in 1976 also had no associated adverse human health effects from radiation. However, all incidents have had significant impact on psychological health and the environment.

Radiation-Induced Injuries and Illnesses

Internal Contamination

After a nuclear detonation or NPP incident, there may be radioactive fallout that can become widespread. This fallout may externally and internally contaminate (NPP only) both casualties and persons within the plume. External decontamination is accomplished through removing the contaminant by either wet or dry means, depending on the chemical composition. Internal contamination may be removed from the body or blocked to the organ of uptake. This should be done urgently, to avoid dose to organs/tissues. Knowledge of this for the oncologist is important as the internally contaminated patient may require therapy for a prolonged amount of time, i.e., after reaching definitive medical care. Following an R/N incident, consult the Radiation Emergency Assistance Center/Training Site by phone (24/7 emergency number, 865-576-1005) or access the REAC/TS RadMed Application, to assist with management of internal contamination with radioactive materials [3].

Acute Radiation Syndrome

Ionizing radiation has many mechanisms that induce injuries and illnesses. These mechanisms may cause direct or indirect injuries. For example, the energy from one ionization event is sufficient to directly break the chemical bonds of macromolecules, including DNA. Ionizing radiation generates free radicals, causing oxidation and oxidative stress, lipid peroxidation, and nitrosative stress, all indirectly injuring molecules and cells. The damage, indirect or direct, to DNA results in either single-strand or double-strand breaks, with the latter often misrejoining, leading to genetic changes including cell death. All of these processes, particularly in the more radiosensitive cells, will manifest as cellular, tissue, vascular, and, ultimately, organ damage. The noxious stimuli of radiation will cause nausea, protracted emesis, and, sometimes, fever as a prodrome. This usually lasts for ~ 48 h; however, higher acute exposures will have a shortened or absent prodrome.

As the dose of the exposure increases, the deterministic effects are seen in their classic syndrome, known as acute radiation syndrome (ARS). ARS is broken down into organ syndromes according to classic presentation at the organ's dose threshold. The earliest observations on the effect of acute radiation exposure on the various organs occurred after the atomic weapon detonations in Japan in August of 1945. In work starting in 1997, the Commission of the European Communities initiated a "Concerted Action" to give a scientific basis and synopsis for deciding the most appropriate medical interventions for ARS. Depending on radiation dose, clinical findings assigned to an organ system may occur concurrently or sequentially with those assigned to the other systems. The signs and symptoms of each of the resulting four organ syndromes of ARS (hematopoietic, gastrointestinal, cutaneous, and neurovascular) are summarized in tables below. The severity of signs and symptoms for each organ system is quantified as degrees of toxicity (degree 1, 2, 3, or 4). The response category (grade 1, 2, 3, or 4) correlates with overall severity of ARS and is determined by the highest degree of toxicity within any of the organ systems [4].

The hematopoietic syndrome (HS) shows classic expression with an acute, whole body (or a large portion of the body) exposure greater than 2 Gray (Gy). This is the identified triage decision point for US guidelines, though other countries use equal to or greater than 3 Gy. The mature lymphocytes are the most sensitive cell type to ionizing radiation. This makes this a helpful tool for triage. Most individuals will fully recover their bone marrow at the 2 Gy level. However, colony-stimulating factors (CSFs) are recommended to more quickly recover the bone marrow and avoid morbidity from infection. The classic HS is lymphopenia, followed by neutropenia, thrombocytopenia, and, eventually, pancytopenia with exposure to greater than 2 Gy. The nadir of neutropenia is generally 3–4 weeks with a classic HS.

The gastrointestinal syndrome (GS) shows classic expression with an acute exposure to the whole/partial body, to include the torso, in the range of 6–8 Gy. The presentation is nausea, protracted emesis, diarrhea, gastrointestinal bleeding, bacterial translocation, and loss of absorptive capability of the intestines. Frank bowel necrosis may occur in the upper range of exposed dose. Acute renal dysfunction/failure may accompany GS.

The cutaneous syndrome (CS) classically begins expression with exposure to 6 Gy or higher. This syndrome presents with erythema, edema, desquamation, and, at higher doses, blistering, ulceration, and necrosis. At the dose threshold for deep ulceration and necrosis, these become complex and often non-healing or recurring wounds.

The neurovascular or cerebrovascular syndrome (NS) presents with acute exposure to doses of 10 Gy and up. This syndrome may present with nausea and protracted emesis within minutes, alteration of mental status or loss of consciousness, ataxia, high fever, fluid shifts and third spacing, seizures, cerebral edema, and coma (especially above 10 Gy). Neurovascular syndrome is not compatible with long-term survivability. Acute renal failure and pulmonary involvement may be concurrent. In acute doses in the 8–30 Gy range, pulmonary edema, pneumonitis, and fibrosis may occur.

Clinical Evaluation of Acute Radiation Injury

Clinical History and Laboratory

Proper application of a well-structured interview technique can contribute to an accurate diagnosis of radiation injury. Patients presenting with nausea, vomiting, and diarrhea may have many etiologies and may be misdiagnosed unless seen in conjunction with a known missing radiological source or incident. Evaluation of a patient suspected of a radiological exposure consists of a thorough event history, including geographic location at the time of the incident; how long the patient was in that location; was the patient sheltered and, if so, what kind of structure and location (basement, windows, etc.); if they exited the area, what pathway was taken and what form of transport was utilized; have they removed clothing/showered; did they have any signs of symptoms; a complete medical/surgical history; and a thorough physical examination. It would also be helpful to know if anyone else was co-located with them and, if so, if that individual is exhibiting similar signs and symptoms.

Laboratory evaluation should include a baseline and serial complete blood cell count with differential (CBC with diff) every 6–12 h to monitor for a decline in the absolute lympho-

 Table 42.1
 Absolute lymphocyte count decrease and approximate

 estimate of absorbed dose (R.E. Goans, personal data, 2014)

	Rough estimate of absorbed
Absolute count 8–12 h post-event	dose
1700–2500/mm ³	0–4 Gy
1200–1700/mm ³	4–8 Gy
<1000/mm ³	>8 Gy
Absolute lymphocyte count 48 h	Absorbed dose estimate
postexposure	
1000–1500/mm ³	1–2 Gy
500-1000/mm ³	2–4 Gy
100–500/mm ³	4–8 Gy
<100/mm ³	>8 Gy

A whole-body dose of 1 Gy or less should not noticeably depress the lymphocyte count below the normal range taken as 1500–3500/mm³

cyte count. In papers by Goans et al., a simple prediction algorithm was presented to estimate effective whole-body dose within 8–12 h after moderate- and high-level gamma accidents and after criticality accidents [5–7]. The algorithm is based on the observation that lymphocyte depletion follows first-order kinetics after high-level gamma accidents. Using historical data from both gamma and criticality accidents, lymphocytes are observed to follow approximately an exponential decline in time within the first 24–48 h. This algorithm has been incorporated into the Armed Forces Radiobiology Research Institute (AFRRI) Biodosimetry Assessment Tool (BAT) program and the WinFrat tool [8] (Table 42.1).

An additional tube (green top, preferred lithium heparin) of blood should be collected for biodosimetry. This tube does not need refrigeration or centrifugation. This tube can be held for several days before being sent to the Radiation Emergency Assistance Center/Training Site (REAC/TS) Cytogenetic Biodosimetry Laboratory (CBL) for dicentric chromosome analysis (see Biodosimetry). A serum amylase may be helpful, as well as a C-reactive protein (CRP) which may be grossly elevated after a whole body exposure.

Blood typing and cross-match may be delayed unless needed for other medical/surgical indications. However, human leukocyte antigen (HLA) typing should be done immediately if the patient is a potential candidate for bone marrow transplantation (see National Network for Management of Mass Radiation Casualties), as it may take time to find a donor match.

Clinical Prodrome

Observation for any of the signs and symptoms of organ system involvement of ARS should be done for smaller incidents. In a mass casualty, patients presenting greater than 4–6 h after the incident who are symptom-free will be triaged as delayed and may follow up later with a community

reception center (CRC), private healthcare provider, or public health point of dispensing (POD). An early sign of significant, potentially non-survivable ARS is early, protracted emesis, not to be confused with a single episode from psychosomatic or sympathetic emesis. Anti-emetics, 5-HT3 receptor antagonists (ondansetron, granisetron), and dopamine antagonist (metoclopramide) are indicated for relief of emesis [4].

In work performed at Oak Ridge Associated Universities (ORAU) from 1964 to 1975, with patients undergoing longterm radiation therapy at a relatively low-dose rate (n=502)patients, 0.8-90 R/h), 50 percentile frequency doses were obtained as follows: ED50=1.08 Gy for anorexia, ED50=1.58 Gy for nausea, and ED50=2.40 Gy for emesis. A trend is noted whereby the time to emesis decreases with increasing dose [9, 10] though there is much variability among individuals and circumstances using this as a sole biodosimeter. The presence of early nausea, protracted emesis, diarrhea, and fever may correlate with the general range of exposure dose, approximately 100% of patients with whole-body dose greater than the LD50 or the dose required to cause mortality in 50% of the population (approximately 3.5-4.0 Gy without treatment) [11]. In addition, Hartmann et al. have noted an increased body temperature for effective whole-body dose >2.5 Gy and acute diarrhea for dose >9 Gy [12]. Other organ dysfunction clinical indicators that may be present up to 4 weeks or later after an terrorist or other incident include:

- Hematologic: lymphopenia, neutropenia, thrombocytopenia with resulting immunocompromise, petechial hemorrhages, and/or mucosal bleeding
- Gastrointestinal: emesis, diarrhea with or without bleeding, fluid loss with dehydration, subsequent renal dysfunction, and potential cardiovascular dysfunction
- Neurovascular or cerebrovascular: mental status changes, cerebral edema, possible seizures, and coma

Cytogenetic Biodosimetry

Since 1963, the dicentric chromosome assay (DCA) has been extensively developed, researched, and validated in numerous incidents to be the global gold standard for biodosimetry [13]. Researchers at AFRRI and REAC/TS have established the conventional lymphocyte metaphase-spread dicentric assay and have applied it to the clinical management of several overexposure incidents. The DCA is radiation-specific, is sensitive as low as 0.2 Gy, and is not affected by ethnicity or gender. REAC/ TS is automating much of the process, and Columbia University has a completely automated system to perform the DCA. One of the limitations of DCA is that it is not available as a point-of-care (POC) test and may take several days for results. Another limitation is the upper end of dose assessment is approximately 5 Gy. Techniques such as premature chromosome condensation (PCC) assay are a good addition to DCA in extending the dose range above the NS syndrome.

Recently, it was suggested that the dicentric assay may be adapted for the triage of mass casualties [14–16]. Lloyd et al. described an in vivo simulation of an accident with mass casualties receiving whole- or partial-body irradiation in the 0–8 Gy range [14]. Faced with an urgent need for rapid results, clinical triage was accomplished by scoring as low as 20 metaphase spreads per subject, compared with the typical 500–1000 spreads scored in routine analyses for estimating dose. However, Lloyd et al. suggested increasing the analyses to 50 metaphase spreads when there is disagreement with the initial assessment or when there is evidence of significant inhomogeneous exposure [14, 17] (Table 42.2).

There are other biodosimetric techniques being researched with point-of-care testing potential. Many of these look for metabolomics and show promise for the future.

Dose range	Proposed validated dosimetry			
(Gy)	method	Prodromal effects	Manifest symptoms	Survival expectancy
0.1–1	Dicentric/PCC	None to mild (1–48 h)	None to slight decrease in blood count	Almost certain
1.0–3.5	Lymphocyte depletion kinetics/ dicentrics/PCC	Mild to moderate (1–48 h)	Mild to severe bone marrow damage	0–10% death
3.5–7.5	Lymphocyte depletion kinetics/PCC	Severe (1–48 h)	Pancytopenia, mild to moderate GI damage	10–100% death within 2–6 weeks
7.5–10.0	Lymphocyte depletion kinetics/PCC	Severe (<1-48 h)	Combined BM and GI damage	90–100% death within 1–3 weeks
>10.0	PCC	Severe (minutes to <48 h)	GI, neurological, cardiovascular damage	100% death (within 2–12 days)

Table 42.2 Proposed biodosimetry technique as a function of expected dos	Table 42.2	Proposed biodosimetry	technique as a	function of	expected dos
--	-------------------	-----------------------	----------------	-------------	--------------

From Prasanna et al. with permission [17]

PCC premature chromosome concentration

Acute Radiation Syndrome (ARS)

The timing and severity of ARS depend upon the radiation quality, dose, dose rate, and irradiated area(s) (organs). Clinical findings associated with an organ system may occur concurrently or sequentially with those assigned to the other systems.

Hematopoietic Syndrome (HS)

Hematopoietic stem/progenitor cells in the bone marrow and circulation are particularly sensitive to ionizing radiation with a dose (D0) of approximately 1 Gy at a dose rate of 0.8 Gy/ min [18] (Table 42.3). At doses of 2-3 Gy, hematopoietic stem/progenitor cells exhibit reduced capacity to divide. Morphological changes in interphase cells of the bone marrow include nuclear karyorrhexis, cytoplasmic fragments, nuclear and intercellular bridging, multinuclearity, and pseudo-Pelger-Huet anomaly [19, 20]. Chromosomal bridges and fragments are seen in actively dividing cells of the marrow. Bone marrow hypoplasia and/or aplasia may develop at doses >5-7 Gy, resulting in severe pancytopenia weeks to months after exposure [21]. The pathophysiologic mechanisms underlying these radiation-induced effects on the bone marrow involve dosedependent and clonal elimination of stem/progenitor cell populations and their progeny [22, 23]. Various degrees of pancytopenia develop several weeks after exposure and need serial CBC and differential every 12-24 hours, as opposed to the more frequent monitoring of the early lymphopenia in the first 24–48 hours post incident [24, 25].

Table 42.3	Hematopoietic syn	drome modified f	from Flied	lner, et al.	[4]

	Degree 1	Degree 2	Degree 3	Degree 4
	Mild	Moderate	Severe	Fatal
	damage	damage	damage	damage
Hematopoietic	system			
Lymphocyte count ^a	At/above 1.5×10^9 cells/L	1.0– 1.5 ×10 ⁹ cells/L	0.5– 1.0 ×10 ⁹ cells/L	Below 0.5×10^9 cells/L
Granulocyte count ^b	At or above 2×10^9 cells/L	1–2 ×10 ⁹ cells/L	0.5– 1.0 ×10 ⁹ cells/L	Below 0.5×10^9 cells/L
Thrombocyte count ^c	At or above 100×10^9 cells/L	50– 100 ×10 ⁹ cells/L	20–50 ×10 ⁹ cells/L	Below 20×10^9 cells/L
Hemoglobin	Normal	Decreases 0–9%	Decreases 10–20%	Decreases more than

Note: Reference values will vary by laboratory

Values are in SI units

 $10^{9}/L = 10^{9}$ cells/ $L = 10^{3}$ cells/ μ L = 10^{3} cells/mm³ (cubic millimeter) ^aReference value: (1.5–4) × 10^{9} cells/L

^bReference value: $(4-9) \times 10^9$ cells/L

^cReference value: $(140-400) \times 10^{9}$ cells/L

Lymphocytes are the most radiosensitive of the circulating blood cells in spite of their being terminally differentiated and largely mitotically inactive. Radiation may alter recirculation properties and surface antigen expression of lymphocytes [4, 26]. The rate of decline in lymphocytes is directly dependent on the absorbed radiation dose.

Other hematological findings include a decline in the absolute neutrophil count (ANC) and the platelet count. The ANC may briefly increase within hours after exposure, a phenomenon first described by Fliedner as an abortive rise [4]. The abortive rise is believed to be due to migration of preformed myeloid elements across the marrow-blood barrier into the circulation, although demargination cannot be excluded as a mechanism for this transient effect. Thereafter, the ANC declines over several days to weeks, depending on radiation dose. The abortive rise is typically seen with HS-1 and HS-2 and appears to indicate reversible marrow damage from a survivable exposure. The absence of an abortive rise in ANC is observed in HS-3 and HS-4 and appears to indicate irreversible bone marrow damage. Neutropenia and thrombocytopenia reach a nadir at 1-2 weeks after exposure to >3-4 Gy. Anemia follows due to impaired erythropoiesis and severe bleeding may occur secondary to thrombocytopenia.

The most significant consequences of lymphopenia and neutropenia are disruption of immune defenses and predisposition to life-threatening infections. ANCs of <500–1000 cells/mm³ (HS-3 and HS-4) are associated with bacterial, viral, and fungal infections, similar to what occurs in the setting of neutropenia and lymphopenia from any other cause. Management of febrile neutropenia should follow guidelines recommended by the Infectious Diseases Society of America (IDSA), using broad-spectrum prophylactic and therapeutic antimicrobial agents [27].

Managing HS includes administration of granulocyte colony-stimulating factor (G-CSF) or granulocytemacrophage colony-stimulating factor (GM-CSF) when the dose is expected to be >2 Gy and/or when it is anticipated that the ANC will decline to <500 cells/mm³ for 7 days or longer [28]. The Food and Drug Administration (FDA) has approved these myeloid colony-stimulating factors for use in a radiological incident. Initiate CSF therapy within 24 h of exposure and continue until the ANC reaches and maintains a level of >1000 cells/mm³ in the absence of active infection. For individuals with active infection, continue cytokines together with antimicrobial agents, according to guidelines of the IDSA [27].

Erythroid-stimulating agents (ESAs) may be used as clinically indicated [28]. The rationale for ESA therapy is avoiding need for red blood cell infusion. The lowest dosage that induces a hemoglobin level of >9–10 g/dL is appropriate. Iron supplementation may also be used in conjunction with ESA therapy. Other growth factors, including stem cell factor, interleukin-3, and the pegylated form of erythropoietin and G-CSF, have been administered sequentially or concomitantly with G-CSF and/or GM-CSF to victims of a radiological incident [28]. These growth factors and new thrombopoietic factors may be useful as in aplastic anemia; however, there is no evidence-based recommendation at this time [29].

Because radiation injury to the bone marrow is typically heterogeneous, leaving areas of unirradiated or minimally irradiated/damaged marrow that are capable of reconstituting lymphohematopoiesis over time, a watch-and-wait approach is recommended after initiating myeloid growth factor therapy. Administration of hematopoietic stem cells (HSCs) should be considered only after failure of a 2-3-week trial of cytokine treatment has been demonstrated [28]. A review of 31 patients undergoing HSC transplantation for accidental radiation injury found that 27 patients died and the remaining 4 patients survived with a rejected allograft [30]. Causes of death after therapeutic HSC transplantation include burns (55%), hemorrhage (41%), infection (15%), and acute respiratory distress syndrome (ARDS) (15%) [31]. Since survival outcomes are poor among HSC transplant recipients with radiation burns, GS, renal failure, and/or adult ARDS, HSC transplantation should not be performed in individuals with non-hematopoietic organ failure and/or active infection [28, 32]. In the case of a large radiological incident, the Radiation Injury Treatment Network (RITN), a voluntary consortium consisting of >70 transplant centers, donor centers, and umbilical cord blood banks, will be activated (see National Network for Management of Mass Radiation Casualties) [33, 34].

When transfusion is indicated for severe cytopenia, blood products should be irradiated (25 Gy) to prevent transfusionassociated graft-versus-host disease (TA-GVHD). Since TA-GVHD is almost universally fatal in this population, its prevention by prior irradiation of blood products is mandatory. Leukoreduction may lessen febrile reactions and the immunosuppressive effects of blood transfusion, limit platelet alloimmunization, and reduce CMV infection [35, 36]. Leukoreduction is recommended whenever feasible.

National Network for Management of Mass Radiation Casualties

In the USA, a network has been developed that includes transplant centers, hospitals, blood donation centers, and stem cell banks to provide resource-intense medical management of mass casualties from a radiological event. The Radiation Injury Treatment Network (RITN) provides comprehensive evaluation and treatment for victims of radiation exposure or other marrow toxic injuries (like those caused by mustard agent). Many of the casualties with radiation injury will be salvageable but require specialized outpatient and/or inpatient care. Recognizing this need, the US National Marrow Donor Program/Be The Match Marrow Registry, the US Navy, and the American Society for Transplantation and Cellular Therapy (formerly known as American Society for Blood and Marrow Transplantation) collaboratively organized RITN, which provides expertise in the management of bone marrow failure, blood component therapy, stem cell collection, and umbilical cord blood banking around the globe.

The RITN is preparing for the resulting medical surge of radiation casualties from the detonation of an improvised nuclear device.

The goals of RITN are:

- To develop treatment guidelines for managing hematologic toxicity among victims of radiation exposure
- To educate healthcare professionals about pertinent aspects of radiation exposure management through training and exercises
- To help coordinate the medical response to radiation events
- To provide comprehensive evaluation and treatment for victims at participating centers

The RITN collaborates with the Department of Health and Human Services-Assistant Secretary for Preparedness and Response to ensure coordination following a mass casualty marrow toxic incident that would require their involvement in a national response. The RITN has developed ARS Treatment Guidelines, Referral Guidelines, as well as webbased training materials for hospitals that receive individuals who show early signs of ARS [37]. In addition, the RITN regularly collaborates with organizations to expand the availability of treatment materials. Through a collaboration with Epic Software Systems, RITN developed Adult and Pediatric Medical Order Sets to be available through the Epic electronic health record system for any hospital that utilizes the system. In collaboration with staff managing the Radiation Emergency Medical Management (REMM) website, RITN developed treatment orders for adults and children [38], and RITN has ongoing efforts with the American Burn Association to formalize combined practice guidelines to meet the needs of caring for patients with combined injuries of trauma and radiation.

The RITN estimates that, of survivors from an IND detonation, only 1% of radiation casualties will be candidates for hematopoietic stem cell transplantation. Approximately 30% of casualties are expected to require specialized supportive care in an inpatient setting that involves isolation to protect individuals with febrile neutropenia. Finally, nearly 70% of casualties are expected to require ambulatory care for treatments such as administration of cytokines and antimicrobials, serial assessment of the CBC, and calculation of the absolute lymphocyte count [37]. RITN medical staff are specialists in hematology and oncologic who have daily experience in treating patients with hematologic signs and symptoms that characterize HS. Hospitals that participate in RITN have established standard operating procedures (SOPs) for managing mass casualties. They coordinate locally with emergency management personnel and public health officials and conduct annual training and exercises to constantly improve their level of preparedness.

Cutaneous Subsyndrome (CS)

Injury to the skin and subcutaneous tissues is highly dependent on localized radiation dose [39-41] (Table 42.4). The external dose thresholds at which the deterministic effect manifest begin at 6 Gy with erythema. A transient epilation may occur at Gy but permanent epilation occurs around the Gy threshold. Desquamation, both dry and moist, will occur at ranges of 10–15 and 15–20 Gy, respectively. At doses greater than 25 Gy, ulceration of deeper tissues and necrosis may occur. Prodromal pain, erythema, and edema may occur for 24–48 h. At higher doses, the prodrome will be shorter or absent. The damage to subcutaneous tissues is highly dependent upon the type and energy of the radiation, as well as the duration of irradiation. These wounds will differ from thermal burns in the pathophysiology. The repeated release of proinflammatory mediators will result in a wound that evolves over weeks, months, or even years. The complex wounds seen at necrosis threshold ranges will often heal with repeated recurrence or not heal completely, often resulting in surgeries and possible amputation. It is important for the hematologistoncologist to recognize that, when the involved total body surface area is greater than 40%, these may cause great morbidity and mortality. Of the 28 acute deaths in the Chernobyl incident, the primary cause of death for 16 of these patients was CS, according to Gottlöber et al. [42]. It is of extreme importance to realize that these cutaneous syndromes and complex

Table 42.4 Cutaneous syndrome modified from Fliedner, et al. [4]

wounds may not fully manifest for 3–4 weeks after an incident, after arrival to a definitive care treatment facility.

Managing CS includes topicals: Class II and III steroids, antihistamines, and antibiotics [4, 43]. The REAC/TS past and present experience has found that hyperbaric oxygen therapy, pentoxifylline with topical and/or oral alphatocopherol, and artificial skin constructs are helpful. Newer silver-based treatments should be considered. Systemic steroids should be decided on a case-by-case basis. Surgical debridement, wide local excision, skin grafts/flaps, or amputation may be necessary [44]. Newer techniques, mesenchymal stem cell [45, 46] or adipose-derived stromal vascular fraction injections [47], with or without surgery, may lead to more long-term healing and relief from intractable pain from compression of cutaneous nerve bundles. Clinical trials using these techniques with thermal burns are underway in the USA. Other countries have both bench research and clinical trials with these therapies; however, lack long-term follow-up is still necessary [45–48].

Gastrointestinal Subsyndrome (GS)

The rapid turnover of the extremely radiosensitive epithelial cells lining the gastrointestinal tract demands that the stem cells in the crypts of Lieberkuhn are functional. The hematologist-oncologist is certainly familiar with the noxious stimuli of various chemotherapy and radiotherapy regimens inducing nausea, anorexia, and, possibly, protracted emesis. This is similar to the prodrome seen with ARS. If nausea and emesis occur within the first hour of exposure and, if accompanied by diarrhea, is likely from both central and peripheral nervous system and is indicative of a potentially non-survivable dose [4]. At doses of 5–6 Gy or greater, damage occurs to stem cells in the crypts, leaving the lining of the gastrointestinal (GI) tract denuded [24, 25]. These findings may be accompanied over time by hematemesis,

	Degree 1	Degree 2	Degree 3	Degree 4
Symptom or sign	Mild damage	Moderate damage	Severe damage	Critical/fatal damage
Cutaneous system				
Erythema ^a	Minimal; transient	Isolated patches 0–9% TBSA ^b	Marked erythema, may have some confluence 10–40% TBSA	Severe; may have confluence >40% TBSA
Pain	None, may have pruritus	Slight	Moderate and persistent	Severe and intractable pain
Blistering and desquamation	Rare, sterile fluid	Rare blister; dry desquamation	Bullae; moist desquamation	Bullae with bleeding; moist desquamation
Deep ulceration/ necrosis		Dermal	Subcutaneous tissue	Muscle, organ, or bone involvement

^aExtent of the TBSA is decisive

^b*TBSA* total body surface area

hematochezia, frank bloody emesis or diarrhea, fluid and electrolyte shifts, hypovolemia with eventual renal failure, and cardiovascular collapse (Table 42.5).

Sloughing of the lining of the GI tract removes the barrier to bacterial translocation from the intestinal lumen to the bloodstream. Bacterial translocation occurs at a time of immunocompromise from neutropenia and lymphopenia, predisposing to sepsis [49].

Managing GS includes antimicrobial prophylaxis and therapy to achieve therapeutic drug levels (rather than bowel decontamination), replacement of fluids and electrolytes, loperamide, and a serotonin receptor antagonist to control emesis [4, 43].

Neurovascular Subsyndrome (NS)

Neurovascular syndrome (NS), characterized by acute, irreversible neurotoxicity, occurs at whole-body doses of >10 Gy (Table 42.6). Abnormalities in EEG may occur after low doses of radiation, and a milder NS may occur at 3–4 Gy, with mild headache, limited emesis, and hypotension. The prodrome begins as noxious stimuli gain access to blood and cerebrospinal fluid and may result in anorexia, nausea, and emesis. The signs and symptoms of moderate- to high-dose exposure include mental status changes, fatigue, dizziness, hypotension, syncope ataxia, seizures, and coma. Managing NS includes serotonin receptor antagonists, mannitol, furosemide, anti-seizure medications, and analgesics [4, 43]. In a mass casualty, resource-scarce setting, these patients would initially be put in an expectant category. Comfort or hospice care should be provided for these patients.

Other Considerations

Involvement of the tracheobronchial tree and lungs is observed at 1-6 months following exposure to a high radiation dose [50]. Edema and leukocyte infiltration of the lung parenchyma occur during the initial day to week after exposure. An acute exudate occurs after 1-3 months, followed by collagen deposition and fibrosis after months to years. Pulmonary edema and acute respiratory distress syndrome (ARDS) may be delayed and seen after high doses [4]. Interstitial pneumonitis accompanied by a restrictive ventilatory defect may lead to death. Management of respiratory failure includes ventilatory support with a lung protection strategy, using the lowest possible inhaled oxygen concentration to maintain an arterial oxygen saturation of >90% [43]. Radiation damage may occur in other organ systems. including the renal, vascular, and cardiac systems. Multiorgan failure (MOF) can be an intermediate- to longterm complication of radiation exposure with significant morbidity and mortality [51, 52]. Vigilance for damage to other organ systems must be maintained throughout medical care.

Table 42.5 Gastrointestinal syndrome modified from Fliedner, et al. [4]

	Degree 1	Degree 2	Degree 3	Degree 4
Symptom or sign	Mild damage	Mild-moderate damage	Severe damage	Serious/fatal damage
Gastrointestinal system				
Diarrhea				
Frequency/	2-3 stools/d; occult	4-6 stools/d; some	7-9 stools/d;	Refractory, watery diarrhea; "gross
bleeding	blood	blood	bloody	hemorrhage"
Mucosal loss/day	Intermittent	Large and intermittent	Persistent	Large, persistent

Fable 42.6	Neurovascular s	yndrome	modified	from	Fliedner,	et al.	[4]	l

	Degree 1	Degree 2	Degree 3	Degree 4
Symptom or sign	Mild damage	Moderate damage	Severe damage	Fatal damage
Neurovascular sys	tem			
Vomiting	Occasional	Intermittent	Persistent	Refractory
Temperature °F/°C	<100.4/38	100.4–104/38–40	>104/40 <24 h	>104/40 >24 h
Hypotension (mm/Hg)	~ normal >100/70	<100/70	<90/60	<80/
Fatigue/ neurologic deficit ^a	Performs activities	Interferes with work/performs activities with noticeable deficit	Assistance for daily activities/cannot perform normal activities; prominent deficit	No activities/loss of consciousness; "life-threatening" neurologic signs
Cognitive deficits	Minor	Moderate	Major impairment since exposure	Complete impairment

^aNeurological deficits: Reflex status including reflexes of the eye, ophthalmoscopy (edema of papilla), fainting, dizziness, ataxia, and other motor signs and sensory signs

Summary

A myriad of specialists, especially those well versed in hematologic abnormalities, will be required for any significant ionizing radiation injury to the whole body, or a significant portion thereof, because of the potential for injury to circulating WBCs or the bone marrow. In fact, inappropriate management of the HS will almost certainly result in elevated morbidity, if not mortality, from the HS itself or damage to other organ systems. The manifestations of immunologic incompetence, including the spectrum of infectious diseases, must be treated properly in order to improve patient survival. Practitioners must be also vigilant for multiple organ dysfunction/failure secondary to ionizing radiation exposure.

Acknowledgments

- Funding sources: ORAU.
- Conflict of interest: None.
- Declarations and disclaimers: This work was performed under Contract # DE-AC05-06OR23100 between Oak Ridge Associated Universities (ORAU) and the US Department of Energy (USDOE). REAC/TS is a program of the Oak Ridge Institute for Science and Education (ORISE), which is operated for the US Department of Energy (DOE) by ORAU. The opinions expressed herein are those of the author and are not necessarily those of the US Government (USG), the US DOE, ORAU, or sponsoring institutions of ORAU. Neither the USG nor the DOE, nor any of their employees, makes any warranty, expressed or implied, or assumes any legal liability or responsibility for the accuracy, completeness, or usefulness of the information contained herein or represents that its use would not infringe on privately owned rights.

REAC/TS, ORISE, US DOE, ORAU, PO Box 117, MS 39, Oak Ridge, TN 37831, USA.

References

- Gollnick DA. Basic radiation protection technology. 6th ed. Altadena: Pacific Radiation Corporation; 2011.
- Browne D, Weiss JF, MacVittie TJ, Pilla MV, editors. Treatment of radiation injuries. New York: Plenum Press; 1990.
- Oak Ridge Institute for Science and Education. REAC/TS RadMed App. https://orise.orau.gov/resources/reacts/radmed-app.html. Accessed 21 Oct 2020.
- 4. Fliedner TM, Friesecke I, Beyrer K, editors. Medical management of radiation accidents: manual on the acute radiation syndrome. London: British Institute of Radiology; 2001.
- Goans RE. Clinical care of the radiation accident patient: patient presentation, assessment, and initial diagnosis. The medical basis for radiation-accident preparedness. The clinical care of victims. In: Proceedings of the fourth international REAC/TS conference

on the medical basis for radiation-accident preparedness. Orlando, FL: The Parthenon Publishing Group; 2002. p. 2001.

- Goans RE, Holloway EC, Berger ME, Ricks RC. Early dose assessment following severe radiation accidents. Health Phys. 1996;72(4):513–8.
- Goans RE, Holloway EC, Berger ME, Ricks RC. Early dose assessment in criticality accidents. Health Phys. 2001;81(4):446–9.
- Google.com/forms. Request to download Armed Forces Radiobiology Research Institute (AFRRI) Biodosimetry Assessment Tool (BAT) program and the WinFrat tool. https://docs. google.com/a/usuhs.edu/forms/d/e/1FAIpQLSc46asLcP3ZrNLg6_ Tnw0ReFmhjWkzg5Hk7IBEIszYuysspZQ/viewform. Accessed 28 Sept 2020.
- Ricks RC, Lushbaugh CC. Studies relative to the radiosensitivity of man: based on retrospective evaluations of therapeutic and accidental total-body irradiation. ORAU/NASA-CR-1444439. 1 Sept 1975.
- Flynn DF, Goans RE. Nuclear terrorism: triage and medical management of radiation and combined-injury casualties. Surg Clin N Am. 2006;86(3):601–36.
- 11. Zhang A. Acute radiation syndrome at a nuclear complex: reconstruction and injury classification of 23 cases from a computer database using the modified Thoma-Wald Triage Model. Masters in Public Health Thesis, University of Pittsburg; 2000.
- Hartmann A, Bojar H, Zamboglou N, Pape H, Schnabel T, Schmitt G. The significance of clinical prodromes for dosage estimation after whole-body radiation exposure. Strahlenther Onkol. 1994;170(9):538–44.
- International Atomic Energy Agency. Cytogenetic analysis for radiation dose assessment: a manual. Technical Reports Series No. 405. Vienna: IAEA; 2001. http://www-pub.iaea.org/MTCD/publications/PDF/TRS405_scr.pdf. Accessed 26 Sept 2020.
- Lloyd DC, Edwards AA, Moquet JE, Guerrero-Carbajal YC. The role of cytogenetics in triage of radiation casualties. Appl Radiat Isot. 2000;52(5):1107–12.
- Voisin P, Benderitter M, Claraz M, Chambrette V, Sorokine-Durm I, Delbos M, et al. The cytogenetic dosimetry of recent accidental overexposure. Cell Mol Biol (Noisy-le-Grand). 2001;47(3):557–64.
- Prassanna PGS, Subramanian U, Greenhill RG, Jacocks JM, Jackson WE, Blakely WF. In proceedings of the 36th health physics society topical meeting. San Antonio: Radiation Safety Aspects of Homeland Security; 2003. p. 218–22.
- Prasanna PGS, Muderhwa JM, Miller AC, Grace MB, Salter CA, Blakely WF. Diagnostic biodosimetry for radiation disasters: current research and service activities at AFRRI. In: NATO medical surveillance and response: research and technology opportunities and options. Neuilly-Sur-Seine. France: North Atlantic Treaty Organization; 2004.
- Fitzgerald TJ, McKenna M, Rothstein L, Daugherty C, Kase K, Greenberger JS. Radiosensitivity of human bone marrow granulocyte-macrophage progenitor cells and stromal colonyforming cells: effect of dose rate. Radiat Res. 1986;107(2):205–15.
- Fliedner TM, Andrews GA, Cronkite EP, Bond VP. Early and late cytologic effects of whole body irradiation on human marrow. Blood. 1964;23:471–87.
- Goans RE, Iddins CJ, Christensen D, Wiley A, Dainiak N. Appearance of pseudo Pelger Huet anomaly after exposure to ionizing in vivo. Health Phys. 2015;108(3):303–7.
- 21. Olivier KR, Peikert T, Owen D. Radiation-induced lung injury. In: Post TW, editor. UpToDate. Waltham MA: UpToDate; 2020.
- 22. Dainiak N. Hematologic consequences of exposure to ionizing radiation. Exp Hematol. 2002;30(6):513–28.
- Dainiak N, Sorba S. Early identification of radiation accident victims for therapy of bone marrow failure. Stem Cells. 1997;15(Suppl 2):275–85.

- Dainiak N, Waselenko JK, Armitage JO, MacVittie TJ, Farese AM. The hematologist and radiation casualties. Hematology Am Soc Hematol Educ Program. 2003:473–96. https://doi.org/10.1182/ asheducation-2003.1.473.
- 25. Waselenko JK, MacVittie TJ, Blakely WF, Pesik N, Wiley AL, Dickerson WE, et al. Medical management of the acute radiation syndrome: recommendations of the Strategic National Stockpile Radiation Working Group. Ann Intern Med. 2004;140(12):1037–51.
- 26. Fliedner TM, Tibken B, Hofer EP, Paul W. Stem cell responses after radiation exposure: a key to the evaluation and prediction of its effects. Health Phys. 1996;70:787.
- Infectious Disease Society of America. http://www.idsociety.org. Accessed 27 Sept 2020.
- 28. Dainiak N, Gent RN, Carr Z, Schneider R, Bader J, Buglova E, et al. First global consensus for evidence-based management of the hematopoietic syndrome resulting from exposure to ionizing radiation. Disaster Med Public Health Prep. 2011;5:202–12.
- 29. Young NS, Maciejewski JP. Aplastic anemia. In: Hoffman R, Benz EJ, Silberstein LE, Heslop HE, Weitz JI, Anastasi J, Salama ME, editors. Hematology: basic principles and practice. 7th ed. Philadelphia: Elsevier; 2018. p. 394–414.
- Dainiak N, Ricks RC. The evolving role of haematopoietic cell transplantation in radiation: potentials and limitations. Br J Radiol. 2005;27(Suppl 1):169–74.
- Abbott B, Ippoliti C, Bruton J, Neumann J, Whaley R, Champlin R. Antiemetic efficacy of granisetron plus dexamethasone in bone marrow transplant patients receiving chemotherapy and total body irradiation. Bone Marrow Transplant. 1999;23:265.
- Allen JY, Wingard JR, Dainiak N. Management of radiation injury. In: Post TW, editor. UpToDate. UpToDate: Waltham MA; 2020.
- Weinstock DM, Case C Jr, Bader J, Chao NJ, Coleman CN, Hatchett RJ, et al. Radiologic and nuclear events: contingency planning for hematologists/oncologists. Blood. 2008;111(12):5440–5.
- 34. National Marrow Donor Program. Radiation Injury Treatment Network. http://www.ritn.net. Accessed 28 Sept 2020.
- Blajchman MA. Immunomodulation and blood transfusion. Am J Ther. 2002;9(5):389–95.
- 36. Hébert PC, Fergusson D, Blajchman MA, Wells GA, Kmetic A, Coyle D, et al.; Leukoreduction Study Investigators. Clinical outcomes following institution of the Canadian universal leukoreduction program for red blood cell transfusions. JAMA 2003;289(15):1941–9.
- National Marrow Donor Program. Radiation Injury Treatment Network. Treatment. http://www.ritn.net/treatment. Accessed 19 Oct 2020.
- U.S. Department of Health & Human Services. National Library of Medicine. REMM Radiation Emergency Medical Management. http://remm.nlm.gov. Accessed 28 Sept 2020.
- 39. Rezvani M, Hopewell JW, Wilkinson JH, Bray S, Morris GM, Charles MW. Time- and dose-related changes in the thickness of skin in the pig after irradiation with single doses of thulium-170 beta particles. Radiat Res. 2000;153(1):104–9.

- Peter RU. Cutaneous radiation syndrome: clinical and therapeutic aspects. Radiol Protect Bull. 1996;183:19–26.
- 41. Meineke V. The role of damage to the cutaneous system in radiationinduced multi-organ failure. Br J Radiol. 2005;27(Suppl 1):85–99.
- 42. Gottlöber P, Steinert M, Weiss M, Bebeshko V, Belyi D, Nadejina N. The outcome of local radiation injuries: 14 years of follow-up after the Chernobyl accident. Radiat Res. 2001;155(3):409–16.
- 43. Dainiak N, Gent RN, Carr Z, Schneider R, Bader J, Buglova E, et al. Literature review and global consensus on management of acute radiation syndrome affecting nonhematopoietic organ systems. Disaster Med Public Health Prep. 2011;5(3):183–201.
- Chambers JA, Purdue GF. Radiation injury and the surgeon. J Am Coll Surg. 2007;204(1):128–39.
- 45. Latillade JJ, Doucet C, Bey E, Carsin H, Huet C, Clairand I, et al. New approach to radiation burn treatment by dosimetry-guided surgery combined with autologous mesenchymal stem cell therapy. Regen Med. 2007;2(5):785–94.
- 46. Akita S, Yoshimoto H, Akino K, Ohtsuru A, Hayashida K, Hirano A. Mesenchymal stem cell therapy in local radiation injuries: a Japanese approach. In: Christensen DM, editor. The medical basis for radiation accident preparedness. Washington, DC: District Creative Printing, Inc.; 2013.
- 47. Iddins CJ, Cohen SR, Goans RE, Wanat R, Jenkins M, Christensen DM, Dainiak N. Case report: industrial x-ray injury treated with non-cultured autologous adipose-derived stromal vascular fraction (SVF). Health Phys. 2016;111(2):112–6.
- 48. Benderitter M, Tamarat R, Prat M, Chapel A, Bottoliers-Depois JF, Lataillade JJ. Regenerative medicine for the medical management of local radiation injury: from bench to bedside. In: Christensen DM, editor. The medical basis for radiation accident preparedness. Washington, DC: District Creative Printing, Inc.; 2013.
- 49. Hauer-Jensen M, Kumar KS, Wang J, Berbee M, Fu Q. Intestinal toxicity in radiation- and combined injury: significance, mechanisms and countermeasures. In: Larche RA, editor. Global terrorism issues and developments. Hauppauge: Nova Science Publishers; 2008.
- 50. Crawford SW. Diagnosis and management of pulmonary problems associated with radiation injury. In: Ricks RC, Berger ME, editors. The medical basis for radiation-accident preparedness: the clinical care of victims. Proceedings of the fourth international REAC/TS conference on medical basis for radiation-accident preparedness. O'Raton, FL: Parthenon Publishing Group; 2002. p. 131–8.
- Gourmelon P, Marquette C, Agay D, Mathieu J, Clarençon D. Involvement of the central nervous system in radiation-induced multi-organ dysfunction and/or failure. Br J Radiol. 2005;78(Suppl 27):62–8.
- Jackson WL, Gallagher C, Myhand RC, Waselenko JK. Medical management of patients with multiple organ dysfunction arising from acute radiation syndrome. Br J Radiol. 2005;78(Supp 27):161–8.



43

Zahra Alhajji and Ahmed Elsayem

Case Study

A 65-year-old woman with advanced uterine cancer presented to the emergency department (ED) with increasing abdominal pain, lethargy, and insomnia. She received her last round of chemotherapy treatment 5 days earlier and takes morphine 15 mg every 4 hours as needed for pain. She had taken the maximum dose for the previous 3 days and was unable to sleep because of increasing pain and anxiety. Her son stated that she was unable to eat or drink (except when taking her pain medication) because of drowsiness. Physical examination demonstrated a chronically ill-appearing patient with dehydration and cachexia. Her abdomen was soft, with generalized tenderness and no rigidity. She was easily arousable and, when asked, reported pain intensity of 10 out of 10. She was tachycardic, but other vital signs were normal. Her son, who was exhausted, left her in the ED and went home. Initial laboratory tests revealed a high blood urea nitrogen level. Intravenous hydration was initiated, and the patient received intravenous morphine 4 mg every hour as needed for severe pain. Shortly after, the patient was taken to radiology for abdominal and head computed tomography. She became agitated, pulled on her IV lines, and began bleeding. She hit the radiology technician, and the ED was notified. The patient was given lorazepam IV to control her agitation, and a 24-hour sitter was ordered. The patient was admitted to the hospital, and palliative care was consulted. Pain medications were changed to fentanyl, and haloperidol was ordered for agitation. Further evaluation revealed cancer progression. A family meeting was held, and the patient was discharged to hospice care.

Z. Alhajji · A. Elsayem (🖂)

Introduction

Delirium is a common neuropsychiatric syndrome among patients with cancer, particularly in elderly patients, those receiving opioids, and those with advanced disease. Over 80% of patients with advanced cancer will develop delirium in the last hours and days before death. Approximately 50% of episodes are reversible [1]. Delirium is a predictor of short survival and is a significant cause of distress among patients, family members, and healthcare providers [2, 3].

Definitions

Terms used in relation to delirium may be imprecise and nonspecific. We will use the following definitions:

- Altered mental status (AMS): A broad term encompasses all manifestations of brain dysfunction, including confusion, clouding of consciousness, disorientation, inattention, altered behavior, or drowsiness [4].
- Confusion: The quality or state of being bewildered or unclear. Confusion is usually a symptom of delirium or dementia [5].
- Delirium: A more specific term commonly used to describe an acute state of confusion resulting from organic brain dysfunction. Delirium is characterized by an acute onset of fluctuating levels of consciousness and severe disturbance of mental abilities which results in confused thinking and reduced awareness of the environment [5–8].

Epidemiology

The exact prevalence of delirium in ED patients with advanced cancer is unknown. We found a frequency of 9% among patients with advanced cancer presenting to the MD Anderson Cancer Center ED [9]. The literature supports rates of 8% to 17% among all elderly patients seen in general EDs

Delirium

Department of Emergency Medicine, The University of Texas MD Anderson Cancer Center, Houston, TX, USA e-mail: aelsayem@mdanderson.org

[10–12]. Among emergently transferred and hospitalized nursing-home elderly, 38% will develop delirium [12]. The highest rates of delirium are seen near the end of life; in one study, 88% of patients with advanced cancer treated in an inpatient palliative care unit experienced terminal delirium. Among hospitalized patients with cognitive impairment, such as dementia, almost half will develop delirium [13]. Siddiqi et al. conducted a systematic review of 42 cohorts and found the prevalence of delirium at admission ranged from 10% to 31%, with the incidence per admission ranging from 3% to 29% and occurrence rates per admission from 11% to 42% [14]. In another systematic review, Gibb et al. reported a delirium frequency of 23% in hospitalized patients [15].

Delirium is even more prevalent among patients with advanced cancer, particularly patients receiving palliative care [16, 17]. In a systematic review of 42 studies, Wattet et al. reported a prevalence ranging from 44% to 88% across all palliative care settings [16]. Fann et al. reported a 50% cumulative incidence of delirium over 30 days following hematopoietic stem cell transplant [18].

Etiology and Pathophysiology

The precise cause of delirium is unknown, but the syndrome tends to occur in predisposed individuals who have underlying comorbid conditions and experience additional insults [1, 19, 20]. Evidence suggests that delirium is a multifactorial condition resulting from structural or nonstructural brain pathology.

Doriathet et al. reported that toxic metabolic factors (frequently multifactorial) are the most common causes of delirium [21]. The most common contributing factors are medications, infection, electrolyte abnormalities, renal insufficiency, liver dysfunction, anticancer drugs, infections, and opioids [22–24].

Tuma and DeAngelis reported that the causes of delirium were multiple in about two-thirds of cases. Only 15% had structural lesions as a cause of AMS [25]. Doriathet et al. studied 100 consecutive patients with confusion, finding that 36% were due to structural causes [21]. Other overlooked but important factors are loneliness and unfamiliar environments, which add to the psychological stress afflicting cancer patients. Given the complex nature of delirium and its multifactorial etiologies, it is difficult to subsume delirium under one heading.

Table 43.1 lists conditions that can cause or contribute to delirium. Medications, particularly opioids, are a frequent proximal cause of delirium in patients with advanced cancer [26, 27]. Table 43.2 lists medications that can contribute to delirium.

Neurotoxicity is a known complication of chronic opioid therapy and may be manifested by delirium, hallucinations Table 43.1 Common conditions that could contribute to delirium

Conditions that	
contribute to delirium	Examples
Medications	Mainly opioids, benzodiazepines, steroids, some antiemetics, certain chemotherapies
Infection	Urinary tract infection, pneumonia, sepsis
Organ failure	Renal, hepatic, pulmonary, cardiac
Primary CNS insult	Primary or metastatic brain tumors, cerebrovascular accidents, leptomeningeal disease
Withdrawal from medications	Alcohol, benzodiazepines
Electrolyte imbalance	Hyponatremia, hypercalcemia
Metabolic/endocrine	Hypothyroidism, paraneoplastic syndrome,
Others	Uncontrolled pain, dehydration, malnutrition, change in environment/ stimulation

 Table 43.2
 Common medications that could contribute to delirium

Medication category	Examples
Opioids analgesia	Morphine, hydromorphone, oxycodone, fentanyl
Antiemetics	Prochlorperazine, promethazine
Antibiotics	Fluoroquinolones; gemifloxacin, levofloxacin, moxifloxacin
Antivirals	Acyclovir
Sedatives	Benzodiazepines; lorazepam, diazepam, midazolam
Muscle relaxants	Baclofen,
Antihistamines	Diphenhydramine, dimenhydrinate, hydroxyzine
Antidepressants and mood stabilizer	TCAs, SSRIs, lithium
Cardiac medications	Antiarrhythmics, digitalis
Chemotherapy	Ifosfamide, 5-fluorouracil, paclitaxel, vincristine, cisplatin, cytosine arabinoside

TCA tricyclic antidepressant, SSRI selective serotonin reuptake inhibitor

(particularly visual hallucinations), myoclonus, and hyperalgesia. Chronic high-dose opioid therapy and coadministration of opioids and other psychoactive drugs, such as benzodiazepines, are common predisposing factors for delirium. Opioidinduced neurotoxicity is believed to be caused by opioid end products, such as morphine-3-glucuronide (in the case of morphine) [23, 28, 29]. The majority of opioids have similar active metabolites. These opioid metabolites are usually excreted by the kidneys and accumulate with renal insufficiency; therefore, dehydration and acute or chronic renal failure may decrease their clearance.

Opioids may exert their toxic effects via the cholinergic system [23]. Central neurotransmitter disturbances are believed to be the final common pathophysiologic mechanism causing delirium. Most of these theories, reviewed by Maldonado, center on neurotransmitter roles, inflammatory cytokines, and blood-brain barrier integrity in delirium development. The neurotransmitter hypothesis builds upon the role of neurotransmitter changes in the development and treatment of delirium and AMS.

Other neurotransmitter abnormalities that may contribute to delirium include excess serotonin and cortisol levels, as well as reductions in GABA (γ -aminobutyric acid) levels [30]. More recently, increasing attention has been given to the role of cytokines, such as interleukins 1 and 6, tumor necrosis factor, and pro-inflammatory markers, such as C-reactive protein, in causing delirium. These are consistent with the neuro-inflammatory hypothesis, which implies that peripheral inflammation results in the activation of CNS parenchymal cells, which produces inflammatory cytokines that alter the normal functioning of neuronal synapses. This hypothesis could explain the "sickness behavior" seen in patients with inflammation or infections [31, 32].

Clinical Personation and Diagnosis

Three clinical subtypes of delirium have been described, classified by the patient's psychomotor activity and arousal level [33]:

- Hyperactive: confusion + agitation ± hallucinations ± delusions ± myoclonus ± hyperalgesia. The patient may fidget around the room, attempt to pull out lines, repeat movements and names, etc. Its features mimic those seen in psychosis, mania, or extrapyramidal side effects of medications.
- Hypoactive: confusion + somnolence ± withdrawal. Usually manifests with withdrawal, psychomotor retardation or lack of movement, lack of orientation, paucity of speech, and unresponsiveness. This type of delirium may mimic a depressed mood.
- Mixed: alternating symptoms of both hyperactive and hypoactive delirium. This form is the most common in advanced cancer patients.

The gold standard for diagnosing delirium is psychiatric evaluation based on criteria set forth by the Diagnostic and Statistical Manual of Mental Disorders (DSM-V). Based on these criteria, diagnosing delirium will require all of the following [34, 35]:

- A. A disturbance in attention (i.e., reduced ability to direct, focus, sustain, and shift attention) and awareness (reduced orientation to the environment).
- B. The disturbance develops over a short period of time (usually hours to a few days), represents a change from baseline attention and awareness, and tends to fluctuate in severity during the course of a day.
- C. An additional disturbance in cognition (e.g., memory deficit, disorientation, language, visuospatial ability, or perception).

- D. The disturbances in Criteria A and C are not explained by another preexisting, established, or evolving neurocognitive disorder and do not occur in the context of a severely reduced level of arousal, such as coma.
- E. There is evidence from the history, physical examination, or laboratory findings that the disturbance is a direct physiological consequence of another medical condition, substance intoxication or withdrawal (i.e., due to a drug of abuse or to a medication), or exposure to a toxin or is due to multiple etiologies.

Patient Evaluation

The clinician should maintain a high index of suspicion and look for clues within the patient's presentation, such as new-onset agitation that peaks at night or variations in sleep patterns (generally sundowning) in which patients sleep during the day and remain awake at night. Also, patients will have memory deficit and difficulty mentioning names of people and places. Relatively acute onset of symptoms with a fluctuating pattern helps distinguish delirium from more chronic conditions such as dementia, which occurs gradually over months and years. Ask the patient specifically about hallucinations (they are more often tactile than visual) and delusional thoughts. Look for clinical signs of sepsis, opioid toxicity, dehydration, metabolic abnormalities, or other potential causes of delirium. Family members or other surrogates may provide valuable information such as changes in cognition, memory, or behaviors. Inattention is a vital finding to observe while assessing delirious patients.

Screening Tools

Multiple tools have been developed in order to facilitate the detection of delirium. Most of these tools used for screening and diagnosis of delirium are based on the DSM criteria. LaMantia conducted a systematic review and concluded that although several tools were used in the ED, only one was validated and, because of limited information, there are no specific recommendations for screening of delirium in ED [36].

The CAM (Confusion Assessment Method) is the most validated screening tool for delirium in the ED [37]. The CAM for the Intensive Care Unit (CAM-ICU) has also been evaluated as an ED screening tool with good reported accuracy [38–40]. In a systematic review, Mariz et al. [41] suggest that the CAM-ICU has potential to be the reference for diagnosing delirium in ED. The Single Question in Delirium is an easy and quick tool to screen for delirium [42].

Table 43.3 lists some commonly used screening and diagnostic tools. Some patients with delirium may appear cog-

Table 43.3 Delirium screening and diagnostic tools

Tool	Purpose	Main Features
Nu-DESC [43]	Screening	Used by nurses at the bedside Brief (5 items, requires <2 min)
MMSE [32]	Screening	Test of cognitive failure 30 items (score <24 signifies greater cognitive failure) Commonly used for dementia but may be used as a screening tool for delirium
CAM [37]	Screening and diagnostic	Based on DSM-III-R [34] criteria Brief (four features, could be completed in <5 min) Simple, but users need training Does not rate the severity
Single Question in Delirium [42]	Screening	Simple and brief Easy to administer
MDAS [44]	Severity rating	Differentiates among subtypes of delirium Easy to administer by a non-medically trained person Ten items, 30-point scale (higher scores signify greater severity) Some groups use cutoff score >7 to diagnose delirium [44]
CAM-ICU [40]	Diagnosis in ICU, screening in ED	Based on CAM Originally validated to assess delirium in ICU Evaluated in ED setting

Nu-DESC Nursing Delirium Screening Scale, *MMSE* Mini-Mental State Examination, *CAM* Confusion Assessment Method, *DSM-III-R* Diagnostic and Statistical Manual of Mental Disorders, Third Edition–Revised, *ED* emergency department, *MDAS* Memorial Delirium Assessment Scale, *CAM-ICU* Confusion Assessment Method for the Intensive Care Unit

nitively normal and follow commands appropriately; thus, using a screening tool can provide valuable guidance toward an accurate diagnosis of the condition. Because of time constraints in the ED, a shorter screening tool may be more practical [43]. The single question delirium screen is a short and practical ED screening tool when answered by the patient and surrogate [34]. However, further research is needed to evaluate its utility. The Memorial Delirium Assessment Scale (MDAS) is easy to administer by a lay-trained person, but it is time-consuming [44]. The MDAS may be more sensitive than CAM in diagnosing ED delirium [45].

Diagnosing delirium superimposed on dementia is challenging. Family identification of delirium using CAM instrument (FAM-CAM) is an informant-based delirium assessment that may help overcome the barriers to delirium screening in patients with dementia [46]. The FAM-CAM consists of 11 items that characterize changes in attention, speech, arousal, and orientation; a patient's family member or caregiver who knows the patient's baseline mental and cognitive status well completes it. To improve delirium recognition in the ED, however, we must move beyond just conducting research and take a more active approach to diagnosing delirium while incorporating implementation science principles.

Differential Diagnosis

Depression and dementia may complicate the differential diagnosis of delirium, and hypoactive delirium may be misdiagnosed as depression, with resultant inappropriate treatment. Agitated delirium may be wrongly assessed as anxiety, insomnia, or psychosis and treated with benzodiazepines. The presence of grimacing or moaning may be interpreted as inadequate pain resulting in inappropriately opioid dosing. Urinary retention and constipation can aggravate agitation. Previously unrecognized cognitive dysfunction may manifest itself as delirium during cancer treatment. Patients may have episodic lucid intervals marked by the clearing of their sensorium.

Management

Step 1. Assessing the Patient

Maintain a high index of suspicion. Ask family members or the nurse caring for the patient about new symptoms, especially symptoms at night.

Use a screening tool such as CAM, MDAS, Clockmaking, or Mini-Mental State Examination (identifies cognitive impairment only). These screening tools should be used even in patients with no overt signs of delirium to make an early diagnosis. Ask specifically about hallucinations (they are more often tactile than visual) and delusional thoughts. Look for clinical signs of sepsis, opioid toxicity, dehydration, metabolic abnormalities, or other potential delirium causes [47, 48].

Order appropriate laboratory tests early, such as a complete blood count, electrolytes, calcium (with albumin), and renal and liver function. Imaging tests that may help include chest x-ray, brain CT/MRI, and O2 saturation, with other tests ordered as clinically indicated.

Step 2. Reviewing the Environment

- Ensure a physically safe environment and minimize noise and excessive light.
- Provide the presence of familiar objects and a visible clock and calendar.
- Include the family by asking them to assist with reorientation.
- Pain levels and the patient's need to use the bathroom should be addressed.

- The bed should be lowered to minimize fall-related injury.
- The patient and surrogate should be advised to call the nurse if the patient wants to leave the bed for any reason, including going to the restroom.
- If no caregiver is available, a sitter should be assigned to assist the patient; however, a familiar face is preferable to a stranger [49, 50].
- Use of restraints or any tethering agent, such as Foley catheters, ECG leads, and oxygen saturation or blood pressure monitor lines, is discouraged. These items may worsen agitation in the delirious patient. Familiar items such as photographs of family members may help calm the patient [47, 51–53].

Step 3. Treating the Symptoms of Delirium

Agitation and Hallucinations Haloperidol is the drug of the first choice. Treatment can be started with haloperidol 0.5 = 1 mg PO/SC/IV q6h and 1–2 mg PO/SC/IV q2h as needed. To bring severe agitation rapidly under control, it may be necessary to give haloperidol more frequently initially (e.g., 1–2 mg q30 min SC/IV as needed during the first hour and q1h prn after that).

The oral bioavailability of haloperidol is approximately 60–70%, and this should be taken into account when con-

verting from the oral to the parenteral route and vice versa. It is essential to bring agitated delirium under control as rapidly as possible to prevent patient, family, and staff distress or injuries. Once symptoms are under control, start reducing the dose to the minimal effective quantity [54, 55]. One of haloperidol's main side effects is extrapyramidal symptoms, which could be confused with worsening agitation [56].

If agitation persists with increasing haloperidol doses, agents providing more sedation, such as chlorpromazine or an atypical antipsychotic (e.g., olanzapine), may be considered. Studies have shown some promise using subcutaneous olanzapine to treat agitated delirium in the advanced cancer patient population [57]. Sometimes an infusion of haloperidol or more sedating neuroleptic medication (such as chlorpromazine) may be required, at which time we encourage palliative care or psychiatric consultation [27, 58, 59]. Table 43.4 lists common medications used in the management of delirium and their main side effects [57, 60, 61].

Chlorpromazine is a more sedating antipsychotic, but its use is limited by significant side effects, such as orthostatic hypotension and anticholinergic effects. Benzodiazepines are not recommended (except in alcohol withdrawal) or if antipsychotic medications are contraindicated or associated with significant side effects (seizures, extrapyramidal side effects, or severe arrhythmia). In such cases, consultation to palliative care, psychiatry, or neurology specialists is recommended [62–64].

Table 43.4 Common medications for management of agitated delirium

Medication	Main receptors affected	Available forms	Major side effects
Haloperidol	Dopamine D2, D1	Oral Subcutaneous Intramuscular Intravenous	Extrapyramidal QT-interval prolongation Seizure *Use lower doses in elderly or frail patients
Chlorpromazine	Dopamine D2, D1 α1-Adrenergic	Oral Rectal Intravenous Intramuscular	Orthostatic hypotension Anticholinergic QT-interval prolongation Arrhythmia
Risperidone	Serotonin 5-HT2A α 1-, α 2- Adrenergic Histamine H1	Oral	Arrhythmia Dysphagia Seizure Expensive *Use with caution in patients with severe hepatic or renal dysfunction Use lower doses in elderly or frail patients
Olanzapine	Serotonin 5HT2A Histamine H1 Muscarinic M1	Oral Subcutaneous Intramuscular	Arrhythmia Weight gain Sedation Expensive *Use with caution in patients with hepatic dysfunction
Quetiapine	Serotonin 5-HT2A Dopamine D2	Oral	Arrhythmia Orthostatic hypotension Expensive
Ziprasidone	Serotonin 5-HT2A Dopamine D2	Oral Intramuscular	Drowsiness Sedation Headache Metabolic disturbances Extrapyramidal symptoms

The atypical antipsychotics have gained attention in the management of delirium; however, most of these agents are unavailable in parenteral form. For atypical antipsychotic agents available in the parenteral form, such as olanzapine and ziprasidone, administration via intramuscular injection is recommended. Such routes may not be suitable for the advanced cancer patient, who frequently suffers from muscle wasting and cachexia or has thrombocytopenia or other hematologic abnormalities. Antipsychotic side effects depend on the receptor targeted or blocked by the medication. For example, chlorpromazine's action on histamine and α 1-adrenergic receptors results in more sedation and orthostatic hypotension, respectively, than does haloperidol. Conversely, haloperidol has more extrapyramidal side effects because of its action at dopamine receptors [63, 65, 66].

Some studies have reported increased mortality in elderly patients receiving atypical antipsychotic medications, resulting from their effect on cardiac conduction. This mortality may be due to prolonged QT intervals and resultant arrhythmias. Therefore, it is vital to obtain a baseline ECG, if possible. If conduction abnormalities are evident at baseline, antipsychotics should be used sparingly.

Step 4. Treating the Underlying Cause

Opioid Toxicity Rotate (switch) opioids from one type to another based on the morphine equivalent dose of the new opioid and reduce the dose of the new drug as recommended for pain management. If the ED physician lacks experience in pain management, consider pain or palliative care consultation. Hydration may be required to clear active metabolites.

Sepsis Start intravenous fluids and appropriate antibiotics as soon as possible.

Other Drugs Discontinue all possible implicated medications, especially benzodiazepines, anticholinergics, corticosteroids, antidepressants, certain antiemetics and antivirals, antibiotics (quinolones), cimetidine, and ranitidine.

Dehydration Start intravenous or subcutaneous hydration as soon as possible. In the home or hospice settings, fluids may be given under the skin (hypodermoclysis) with normal saline at 60–100 ml/hour or give boluses of 500 cc administered over 1 hour, three or four times daily.

Hypercalcemia Treat with hydration with saline and bisphosphonates.

Hypoxia Treat the underlying cause and administer oxygen.

Table 43.5 Interventions to correct common underlying causes of delirium

Condition	Treatment
Side effect of a medication (e.g., opioids, benzodiazepines)	Stop offending drug (with opioids, consider reducing dose/switching to a different opioid)
Alcohol or benzodiazepine withdrawal	Benzodiazepines
Dehydration	Intravenous fluid hydration
Infection/sepsis	Antibiotics
Hypercalcemia	Hydration, bisphosphonate, or calcitonin (for hypercalcemia)
Brain metastasis	Corticosteroids (consider radiation therapy or chemotherapy)

Brain Tumor or Metastasis *C*onsider high-dose corticosteroids. Radiation or surgery may be required. Table 43.5 lists possible interventions to correct the underlying cause of delirium.

Step 5. Counseling the Patient's Family and Healthcare Professionals

Confusion and agitation are expressions of brain malfunction and not necessarily of discomfort or suffering. Disinhibition is one of the main components of delirium and may result in two possibly distressing phenomena:

- A dramatic expression of previously well-controlled physical symptoms by grimacing or moaning: Family or staff may interpret this as aggravating symptoms rather than merely increased expression. In addition to observer distress, this can lead to excessive use of opioids and adjuvant drugs, as well as the accompanying potential for delirium exacerbation.
- Unreasonable requests of family or staff (e.g., "I want to go home now."): If these requests are not immediately addressed, the patient may become hostile. Unless appropriately explained to the patient's family, this disinhibited behavior may be quite distressing. Research has shown that up to 74% of patients recall their symptoms after delirium resolution and up to 81% of those recall associated distress [67]. Hypoactive delirium is as distressing for patients as hyperactive delirium.

Palliative Sedation

Palliative sedation (PS) is defined as "the monitored use of sedative medication to reduce patients' awareness of intractable and refractory symptoms near the end of life, when other interventions have failed to control them." Clinicians should attempt all available symptom-specific measures, including specialists' consultation before labeling a symptom as refractory. Refractory delirium is the most common indication for PS [68].

The goal of PS is to control distressing symptoms and not to hasten death. Consciousness should be maintained if possible. PS should be differentiated from physician-assisted death or euthanasia. Therefore, we do not recommend PS for existential distress. Reasons and goals of PS should be discussed with the family and patient (if possible) and well documented in the medical record [68].

Midazolam is the drug of the first choice. Use the lowest dose possible to provide comfort, e.g., commence a continuous SC or IV infusion of midazolam at 1 mg/hour, titrated according to clinical response.

Prognosis

Mortality risk varies depending on the delirium subtype, as well as potential etiological factors involved. The hypoactive subtype of delirium is associated with a higher mortality rate and a shorter mean survival time [17].

Delirium is associated with increased risks of complications, more extended hospital stays, prolonged mechanical ventilation, and increased mortality rates. Delirium has also been suggested as a risk factor for the future development of dementia. Patients who do not recover fully continue to show signs of cognitive impairment and functional decline over time [69].

References

- Lawlor PG, Gagnon B, Mancini IL, Pereira JL, Hanson J, Suarez-Almazor ME, Bruera ED. Occurrence, causes, and outcome of delirium in patients with advanced cancer: a prospective study. Arch Intern Med. 2000;160(6):786–94.
- Salluh JI, Wang H, Schneider EB, Nagaraja N, Yenokyan G, Damluji A, et al. Outcome of delirium in critically ill patients: systematic review and meta-analysis. BMJ. 2015;350:h2538.
- Witlox J, Eurelings LS, de Jonghe JF, Kalisvaart KJ, Eikelenboom P, van Gool WA. Delirium in elderly patients and the risk of postdischarge mortality, institutionalization, and dementia: a metaanalysis. JAMA. 2010;304(4):443–51.
- Collopy KT, Kivlehan S, Snyder SR. Acute altered mental status in elderly patients: what can cause geriatric AMS and delirium? EMS World. 2013;42(2):31–9.
- Wong N, Abraham G. Managing delirium in the emergency department: tools for targeting underlying etiology. Emerg Med Pract. 2015;17(10):1–20; quiz 21.
- Ohl ICB, Chavaglia SRR, Ohl RIB, Lopes MCBT, Campanharo CRV, Okuno MFP, Batista REA. Evaluation of delirium in aged patients assisted at emergency hospital service. Rev Bras Enferm. 2019;72(Suppl 2):153–60.
- Lee S, Gottlieb M, Mulhausen P, Wilbur J, Reisinger HS, Han JH, Carnahan R. Recognition, prevention, and treatment of delirium in emergency department: an evidence-based narrative review. Am J Emerg Med. 2020;38(2):349–57.

- Milisen K, Lemiengre J, Braes T, Foreman MD. Multicomponent intervention strategies for managing delirium in hospitalized older people: systematic review. J Adv Nurs. 2005;52(1):79–90.
- Elsayem AF, Bruera E, Valentine AD, Warneke CL, Yeung SC, Page VD, et al. Delirium frequency among advanced cancer patients presenting to an emergency department: a prospective, randomized, observational study. Cancer. 2016;122(18):2918–24.
- Han JH, Wilson A, Vasilevskis EE, Shintani A, Schnelle JF, Dittus RS, et al. Diagnosing delirium in older emergency department patients: validity and reliability of the delirium triage screen and the brief confusion assessment method. Ann Emerg Med. 2013;62(5):457–65.
- 11. Ke YT, Peng AC, Shu YM, Chung MH, Tsai KT, Chen PJ, et al. Prevalence of geriatric syndromes and the need for hospice care in older patients of the emergency department: a study in an asian medical center. Emerg Med Int. 2020;2020:7174695.
- Dwyer R, Gabbe B, Stoelwinder JU, Lowthian J. A systematic review of outcomes following emergency transfer to hospital for residents of aged care facilities. Age Ageing. 2014;43(6):759–66.
- Britton A, Russell R. WITHDRAWN: multidisciplinary team interventions for delirium in patients with chronic cognitive impairment. Cochrane Database Syst Rev. 2007;(2):CD000395.
- Siddiqi N, House AO, Holmes JD. Occurrence and outcome of delirium in medical in-patients: a systematic literature review. Age Ageing. 2006;35(4):350–64.
- Gibb K, Seeley A, Quinn T, Siddiqi N, Shenkin S, Rockwood K, Davis D. The consistent burden in published estimates of delirium occurrence in medical inpatients over four decades: a systematic review and meta-analysis study. Age Ageing. 2020;49(3): 352–60.
- Watt CL, Momoli F, Ansari MT, Sikora L, Bush SH, Hosie A, et al. The incidence and prevalence of delirium across palliative care settings: a systematic review. Palliat Med. 2019;33(8):865–77.
- Hosie A, Davidson PM, Agar M, Sanderson CR, Phillips J. Delirium prevalence, incidence, and implications for screening in specialist palliative care inpatient settings: a systematic review. Palliat Med. 2013;27(6):486–98.
- Fann JR, Hubbard RA, Alfano CM, Roth-Roemer S, Katon WJ, Syrjala KL. Pre- and post-transplantation risk factors for delirium onset and severity in patients undergoing hematopoietic stem-cell transplantation. J Clin Oncol. 2011;29(7):895–901.
- Sagawa R, Akechi T, Okuyama T, Uchida M, Furukawa TA. Etiologies of delirium and their relationship to reversibility and motor subtype in cancer patients. Jpn J Clin Oncol. 2009;39(3):175–82.
- Morita T, Tei Y, Tsunoda J, Inoue S, Chihara S. Underlying pathologies and their associations with clinical features in terminal delirium of cancer patients. J Pain Symptom Manag. 2001;22(6):997–1006.
- Doriath V, Paesmans M, Catteau G, Hildebrand J. Acute confusion in patients with systemic cancer. J Neuro-Oncol. 2007;83(3): 285–9.
- Dasgupta M, Hillier LM. Factors associated with prolonged delirium: a systematic review. Int Psychogeriatr. 2010;22(3):373–94.
- Wiffen PJ, Derry S, Moore RA. Impact of morphine, fentanyl, oxycodone or codeine on patient consciousness, appetite and thirst when used to treat cancer pain. Cochrane Database Syst Rev. 2014;2014(5):CD011056.
- 24. Rosen T, Connors S, Clark S, Halpern A, Stern ME, DeWald J, et al. Assessment and management of delirium in older adults in the emergency department: literature review to inform development of a novel clinical protocol. Adv Emerg Nurs J. 2015;37(3):183–96; quiz E3.
- Tuma R, DeAngelis LM. Altered mental status in patients with cancer. Arch Neurol. 2000;57(12):1727–31.
- Talbot-Stern JK, Green T, Royle TJ. Psychiatric manifestations of systemic illness. Emerg Med Clin North Am. 2000;18(2):199–209, vii-viii.

- El Majzoub I, Abunafeesa H, Cheaito R, Cheaito MA, Elsayem AF. Management of altered mental status and delirium in cancer patients. Ann Palliat Med. 2019;8(5):728–39.
- Morita T, Tei Y, Tsunoda J, Inoue S, Chihara S. Increased plasma morphine metabolites in terminally ill cancer patients with delirium: an intra-individual comparison. J Pain Symptom Manag. 2002;23(2):107–13.
- 29. Maldonado JR. Delirium pathophysiology: an updated hypothesis of the etiology of acute brain failure. Int J Geriatr Psychiatry. 2018;33(11):1428–57.
- Yoshitaka S, Egi M, Kanazawa T, Toda Y, Morita K. The association of plasma gamma-aminobutyric acid concentration with postoperative delirium in critically ill patients. Crit Care Resusc. 2014;16(4):269–73.
- Hoogland IC, Houbolt C, van Westerloo DJ, van Gool WA, van de Beek D. Systemic inflammation and microglial activation: systematic review of animal experiments. J Neuroinflammation. 2015;12:114.
- Mattar I, Chan MF, Childs C. Factors causing acute delirium in critically ill adult patients: a systematic review. JBI Libr Syst Rev. 2012;10(3):187–231.
- 33. de Rooij SE, Schuurmans MJ, van der Mast RC, Levi M. Clinical subtypes of delirium and their relevance for daily clinical practice: a systematic review. Int J Geriatr Psychiatry. 2005;20(7):609–15.
- Han JH, Wilson A, Schnelle JF, Dittus RS, Ely EW. An evaluation of single question delirium screening tools in older emergency department patients. Am J Emerg Med. 2018;36(7):1249–52.
- 35. Morandi A, Davis D, Bellelli G, Arora RC, Caplan GA, Kamholz B, et al. The diagnosis of delirium superimposed on dementia: an emerging challenge. J Am Med Dir Assoc. 2017;18(1):12–8.
- LaMantia MA, Messina FC, Hobgood CD, Miller DK. Screening for delirium in the emergency department: a systematic review. Ann Emerg Med. 2014;63(5):551–60.e2.
- 37. Inouye SK, van Dyck CH, Alessi CA, Balkin S, Siegal AP, Horwitz RI. Clarifying confusion: the confusion assessment method. A new method for detection of delirium. Ann Intern Med. 1990;113(12):941–8.
- Han JH, Wilson A, Graves AJ, Shintani A, Schnelle JF, Dittus RS, et al. Validation of the confusion assessment method for the intensive care unit in older emergency department patients. Acad Emerg Med. 2014;21(2):180–7.
- 39. Van de Meeberg EK, Festen S, Kwant M, Georg RR, Izaks GJ, Ter Maaten JC. Improved detection of delirium, implementation and validation of the CAM-ICU in elderly emergency department patients. Eur J Emerg Med. 2017;24(6):411–6.
- 40. Ely EW, Margolin R, Francis J, May L, Truman B, Dittus R, et al. Evaluation of delirium in critically ill patients: validation of the confusion assessment method for the intensive care unit (CAM-ICU). Crit Care Med. 2001;29(7):1370–9.
- Mariz J, Costa Castanho T, Teixeira J, Sousa N, Correia Santos N. Delirium diagnostic and screening instruments in the emergency department: an up-to-date systematic review. Geriatrics (Basel). 2016;1(3):22.
- 42. Sands MB, Dantoc BP, Hartshorn A, Ryan CJ, Lujic S. Single Question in Delirium (SQiD): testing its efficacy against psychiatrist interview, the confusion assessment method and the memorial delirium assessment scale. Palliat Med. 2010;24(6):561–5.
- 43. Gaudreau JD, Gagnon P, Harel F, Tremblay A, Roy MA. Fast, systematic, and continuous delirium assessment in hospitalized patients: the nursing delirium screening scale. J Pain Symptom Manag. 2005;29(4):368–75.
- 44. Breitbart W, Rosenfeld B, Roth A, Smith MJ, Cohen K, Passik S. The memorial delirium assessment scale. J Pain Symptom Manag. 1997;13(3):128–37.
- Elsayem AF, Bruera E, Valentine A, Warneke CL, Wood GL, Yeung SJ, et al. Advance directives, hospitalization, and survival among

advanced cancer patients with delirium presenting to the emergency department: a prospective study. Oncologist. 2017;22(11):1368–73.

- 46. Mailhot T, Darling C, Ela J, Malyuta Y, Inouye SK, Saczynski J. Family identification of delirium in the emergency department in patients with and without dementia: validity of the Family Confusion Assessment Method (FAM-CAM). J Am Geriatr Soc. 2020;68(5):983–90.
- 47. Grossmann FF, Hasemann W, Graber A, Bingisser R, Kressig RW, Nickel CH. Screening, detection and management of delirium in the emergency department - a pilot study on the feasibility of a new algorithm for use in older emergency department patients: the modified Confusion Assessment Method for the Emergency Department (mCAM-ED). Scand J Trauma Resusc Emerg Med. 2014;22:19.
- Arendts G, Love J, Nagree Y, Bruce D, Hare M, Dey I. Rates of delirium diagnosis do not improve with emergency risk screening: results of the Emergency Department Delirium Initiative Trial. J Am Geriatr Soc. 2017;65(8):1810–5.
- 49. Kroll L, Böhning N, Müßigbrodt H, Stahl M, Halkin P, Liehr B, et al. Non-contact monitoring of agitation and use of a sheltering device in patients with dementia in emergency departments: a feasibility study. BMC Psychiatry. 2020;20(1):165.
- 50. Abraha I, Rimland JM, Trotta F, Pierini V, Cruz-Jentoft A, Soiza R, et al. Non-pharmacological interventions to prevent or treat delirium in older patients: clinical practice recommendations The SENATOR-ONTOP Series. J Nutr Health Aging. 2016;20(9):927–36.
- Gual N, Morandi A, Pérez LM, Brítez L, Burbano P, Man F, Inzitari M. Risk factors and outcomes of delirium in older patients admitted to postacute care with and without dementia. Dement Geriatr Cogn Disord. 2018;45(1–2):121–9.
- Sampson EL, West E, Fischer T. Pain and delirium: mechanisms, assessment, and management. Eur Geriatr Med. 2020;11(1):45–52.
- 53. Abraha I, Cruz-Jentoft A, Soiza RL, O'Mahony D, Cherubini A. Evidence of and recommendations for non-pharmacological interventions for common geriatric conditions: the SENATOR-ONTOP systematic review protocol. BMJ Open. 2015;5(1):e007488.
- 54. Tanimukai H, Tsujimoto H, Matsuda Y, Tokoro A, Kanemura S, Watanabe M, et al. Novel therapeutic strategies for delirium in patients with cancer: a preliminary study. Am J Hosp Palliat Care. 2016;33(5):456–62.
- 55. Hui D, Frisbee-Hume S, Wilson A, Dibaj SS, Nguyen T, De La Cruz M, et al. Effect of lorazepam with haloperidol vs haloperidol alone on agitated delirium in patients with advanced cancer receiving palliative care: a randomized clinical trial. JAMA. 2017;318(11): 1047–56.
- Tamune H. Delirium-informed care in emergency departments: diagnosis and beyond. Am J Emerg Med. 2020;38(8):1684–5.
- 57. Elsayem A, Bush SH, Munsell MF, Curry E 3rd, Calderon BB, Paraskevopoulos T, et al. Subcutaneous olanzapine for hyperactive or mixed delirium in patients with advanced cancer: a preliminary study. J Pain Symptom Manag. 2010;40(5):774–82.
- Roth AJ, Breitbart W. Psychiatric emergencies in terminally ill cancer patients. Hematol Oncol Clin North Am. 1996;10(1):235–59.
- Bascom PB, Bordley JL, Lawton AJ. High-dose neuroleptics and neuroleptic rotation for agitated delirium near the end of life. Am J Hosp Palliat Care. 2014;31(8):808–11.
- Boettger S, Breitbart W. An open trial of aripiprazole for the treatment of delirium in hospitalized cancer patients. Palliat Support Care. 2011;9(4):351–7.
- Vodovar D, Malissin I, Deye N, Baud FJ, Mégarbane B. Olanzapine postinjection delirium/sedation syndrome: an unrecognized diagnosis in the emergency department. J Emerg Med. 2014;47(1):e23–4.
- Gonin P, Beysard N, Yersin B, Carron PN. Excited delirium: a systematic review. Acad Emerg Med. 2018;25(5):552–65.
- 63. Dixit D, Shrestha P, Adelman M. Neuroleptic malignant syndrome associated with haloperidol use in critical care setting: should haloperidol still be considered the drug of choice for the man-

agement of delirium in the critical care setting? BMJ Case Rep. 2013;2013:bcr2013010133.

- 64. Latronico N. Haloperidol and delirium in the ICU: the finger pointing to the moon. Intensive Care Med. 2018;44(8):1346–8.
- Gonçalves F, Almeida A, Pereira S. A protocol for the control of agitation in palliative care. Am J Hosp Palliat Care. 2016;33(10):948–51.
- Nauck F, Alt-Epping B. Crises in palliative care--a comprehensive approach. Lancet Oncol. 2008;9(11):1086–91.
- 67. Bruera E, Bush SH, Willey J, Paraskevopoulos T, Li Z, Palmer JL, et al. Impact of delirium and recall on the level of distress in

patients with advanced cancer and their family caregivers. Cancer. 2009;115(9):2004-12.

- Beller EM, van Driel ML, McGregor L, Truong S, Mitchell G. Palliative pharmacological sedation for terminally ill adults. Cochrane Database Syst Rev. 2015;1(1):CD010206.
- Bush SH, Lawlor PG, Ryan K, Centeno C, Lucchesi M, Kanji S, et al. ESMO Guidelines Committee. Delirium in adult cancer patients: ESMO clinical practice guidelines. Ann Oncol. 2018;29(Suppl 4):iv143–65.

Suicide

Anis Rashid



44

Introduction

Suicide is the intentional act of taking one's own life. The traits dependent on heritable factors related to suicidal psychopathology are complex. The triggers leading to suicide vary from severe depression to loss of a loved one, loss of a job, hopelessness, substance misuse, chronic or incurable illness, and even uncontrollable anger due to an underlying clinical or psychiatric illness. According to the World Health Organization, about 800,000 people die by suicide worldwide every year [1]. Suicide is the second leading cause of death in individuals aged 18–24 years and the fourth leading cause of death in older populations [2].

Cancer patients have a four times higher risk of suicide as compared to the general population [3]. One study found that 8% of patients attending an oncology clinic reported ongoing passive suicidal thoughts [4]. Chochinov et al. documented that up to 44% of terminally ill cancer patients expressed fleeting suicidal thoughts [5]. Cancer patients are at high risk for suicide as they are burdened with cancer-specific risk factors in addition to the general risk factors for suicide. Major risk factors specific to cancer patients include pain, physical dysfunction/dependence, and a fear of abandonment [6]. Literature indicates that the risk of suicide is highest in the first year and the relative risk is highest in the first month after a cancer diagnosis but decreases significantly over time [6]. Increased risk may resurface with advanced stage or recurrent disease. A higher risk of suicide has been reported in patients with certain cancer types such as lung cancer, head and neck cancers, Hodgkin lymphoma, testicular cancer, and bladder cancers [3]. Studies have shown that elderly, white, unmarried men are at greater risk than other patient groups [3] and as compared to patients with spousal support [7].

Whether in the general population or in an oncology setting, the act of suicide has a traumatic effect not only on

A. Rashid (🖂)

the person's family, relatives, close friends, and caregivers but also on coworkers, colleagues, organizations the person was part of, and even the entire community. Caregivers of cancer patients may be more traumatized, as in addition to grief they may feel they have failed in providing proper emotional and physical support to the patients [8]. Suicide has repercussions far beyond the affected individual. In summary, the act of suicide has negative ripple effects throughout society. Confidentiality is the key to preservation of trust between clinicians and patients, but when there is a threat of harm to self or others, clinicians have a duty to breach confidentiality and take necessary actions to protect identifiable individuals. Recognizing and reporting suicidal individuals whose job functions could endanger the safety of others is crucial.

A few years ago, one such suicide changed the dynamics of the airline industry, with global repercussions. On March 24, 2015, Germanwings Flight 9525, an Airbus 320, took off for a two-hour flight from Barcelona to Dusseldorf. An hour and half into the flight, the Airbus 320 crashed into the Alps 100 kilometers northwest from Niles killing all 150 people on board. The cause of the crash, determined by an investigation, was an act of suicide committed by the copilot, 27-yearold Andreas Lubitz. The suicide shook the airline industry worldwide. Subsequently, it was determined through an investigation that the copilot had a long history of severe depression and was on several psychotropic medications. He had seen his psychiatrist 2 weeks before the incident. German privacy laws prevented this information from being shared with his employer. This lack of information prevented any timely intervention by his employer or a third party. The result was a terrible suicide that took both his life and that of 149 innocent people. It also brought into focus the question of privacy laws in relation to conditions relevant to critical job functions.

In the United States, the patient physician relationship is confidential, and the Health Insurance Portability and Accountability Act (HIPAA) protects the confidentiality of health records, including therapy and mental health infor-

© Springer Nature Switzerland AG 2021

Department of Psychiatry, The University of Texas MD Anderson Cancer Center, Houston, TX, USA e-mail: anrashid@mdanderson.org

K. H. Todd et al. (eds.), Oncologic Emergency Medicine, https://doi.org/10.1007/978-3-030-67123-5_44

mation, but with certain exceptions. Confidentiality can be breached if the patient is a threat to his/her own safety and safety of others [9].

Despite advances in cancer treatment and success in novel approaches of treatment with immunotherapy, the diagnosis of cancer can still feel like a death sentence to some patients. Patients can feel anxiety, depression, and compromised quality of life due to the sheer burden of undergoing cancer treatment. Regular support, reassurance, and medication management by mental health providers give them hope, improve their quality of life, and keep them adherent to medical treatment. Continuity of care and empowering them with emotion-based coping strategies are essential for their stability.

Case Study: Refractory Hodgkin Lymphoma

Mr. B, a 40-year-old gentleman with refractory Hodgkin lymphoma, was seen over several years in the psychiatry clinic for anxiety and depression. He was stable with medications and psychotherapy. However, Mr. B lost his mental health insurance and was not seen in the clinic for several months. One afternoon a social worker from the lymphoma team contacted the psychiatry clinic, explaining that Mr. B is extremely distressed, is back on mental health coverage, and needs to be seen urgently. Mr. B walked into the psychiatry clinic limping, with his head down, and spoke in a low tone. He stated he was miserable due to CAR T-cell toxicities and feels he is better off being dead. He denied any active suicidal thoughts or plans but reported he cannot live like this and intends to stop the treatment. He came to psychiatry clinic at the social worker's insistence.

Mr. B had failed several lines of chemotherapy and stem cell transplant. The disease progressed, and he was then started on immune checkpoint inhibitor. Mr. B developed severe toxicities related to the treatment which compromised his quality of life. He was fatigued and developed bilateral lower limb vasculitis with open wounds, abdominal pain, diarrhea, bilateral hip pain, neuropathy in hands and feet, and a wrist drop. He became severely depressed and started having fleeting suicidal thoughts. Mr. B reconnected with his psychiatrist and assured her that he would not harm himself because of his faith. As he had no active suicidal thoughts, no past history of suicidal attempt, and a good support system, he did not seem to be at acute risk for suicide. He was discharged home with close follow-up and was given suicide hotline numbers and the social worker's direct contact information. Social work's timely intervention and support from psychiatry services, as well as a modification in his cancer treatment regimen, has stabilized Mr. B. He is back to his baseline mood, compliant with his cancer treatment, and stable enough to enjoy company of his friends and his favorite television programs.

It has been well established that not all individuals who commit suicide truly want to die. Most suicidal patients have a desire to live; they either cannot see past their distress to the options available or have lost hope of changing their situation. It is important to be able to assess who will commit suicide and who is just thinking about it. The relationship between depression and suicide has been well documented in the literature; however, not everyone with depression wants to die or attempts suicide. This suggests that depression is not fully adequate as a causal explanation and that additional factors play a role in suicide. Although the act of suicide is complex, several theories have been described to explain the extent of deep emotional pain and reasoning that triggers the act. In this chapter, we describe these theories and the biopsychosocial model of suicide. We then discuss risk factors specific to cancer patients and management of suicidal patients in oncology settings. Finally, we propose directions for future research.

Theories of Suicide

Mental health providers have been trained to understand the psychopathology underlying suicidal thoughts. Psychoanalysts have several theories regarding suicidal ideations and gestures. According to Freud, suicide is aggression turned inward [10]. He believed that patients internalize and connect with an object (person) so deeply that they and the object become one entity. If in their minds that object betrays them, they become angry and believe that the only way to get rid of the object is to get rid of themselves [10]. Karl Menninger claimed suicide reflects murderous wishes of individuals who kill themselves [11]. Edwin Shneidman described the unbearable psychological pain and suffering that exists in the mind of a suicidal person [12]. He talked about the imbalance between belongings and the burden. He believed that individuals with major losses and failures in life may feel overpowered by psychological demands that are above their tolerance threshold. This intense psychological energy can lead to an impulsive desire to change the unbearable situation [12]. Aaron Beck's theory of suicide is about hopelessness: that nothing will change for the better no matter what an individual does [13]. Another theory of suicide by Roy Baumeister is based on persistent negative experiences in life leading to negative self-perception and self-hate, with the only escape being suicide [14].

Biopsychosocial Model of Suicide

Introduced about 45 years ago, the biopsychosocial model was developed to better understand patients by using a more holistic approach with less emphasis on biomedicine. This model, which captures the interaction of biological, psychological, and social factors underlying illness, can help nonpsychiatric clinicians understand the underlying complex psychopathology that may manifest as physical symptoms in their patients [15]. With time, the biopsychosocial model became very popular and currently has become the standard of practice in teaching programs for medical students and residents.

Unfortunately, the concept and practice of this model have changed overtime and have now drawn an artificial distinction between biology and psychology. Psychiatrists have become psychopharmacologists, and the role of psychotherapy has been taken over by psychologists. One of the reasons for this divide is the difference in reimbursement from insurance companies for pharmacological management versus psychotherapy.

Trained in the concept that any psychopathology may have biological and social influences, mental health providers are highly interested in examining **three** domains: **predisposing, precipitating**, and **perpetuating** risk factors for suicide. Predisposing factors include patients' genetic makeup, temperament, and social factors during early childhood. Precipitating factors include stressful life events and losses affecting psychology. Perpetuating factors are ongoing life challenges [16]. The biopsychosocial formulation gives clinicians a deeper understanding of such factors as patients' unhappy childhoods, poor social support and relationships, anger, negative cognitive style, and impulsivity.

Predisposing Factors: Genetic Makeup

Serotonin

One of the complex aspects of the neurobiology of suicide is that of serotonin. Dysregulation of serotonergic receptors in the brain plays an important role in depression and suicidal behavior. Serotonin is the main neurotransmitter that plays an important role in mood [17], appetite, sleep, and memory [18]. It has been well documented that cerebrospinal fluid in depressed patients who committed suicide contained lower levels of 5-hydroxyindoleacetic acid, which is the major metabolite of serotonin, as well as lower levels of homovanillic acid, a metabolite of dopamine [19]. Lower levels of serotonin could be due to upregulation of serotonin receptors [19]. In order to compensate for lower serotonin levels, the body attempts to enhance serotoninergic activity at nerve terminals by increasing central serotonergic neurons and the levels of key enzymes required for serotonin synthesis. This complex phenomenon is still not clearly understood.

Hypothalamus-Pituitary-Adrenal Axis

A more recent study focused on the hypothalamus-pituitaryadrenal axis in depressed patients and confirmed hyperfunctioning of the axis in depression and return to normal functioning when symptoms improved. The study used the dexamethasone suppression test and found a much higher positive rate in depressed patients with suicidal behavior vs. depressed patients with no previous suicidal thoughts. This indicates suicide-specific characteristics that may differentiate depressed patients with suicidal behavior from those without such behavior [20]. Further work is needed in the area.

Childhood Trauma

Other associated factors affecting suicidal thought include adverse childhood experiences in the absence of protective factors. Children raised in a household with parental substance use or incarceration and divorce or domestic violence or those who experience childhood sexual, emotional, or physical abuse develop a sense of insecurity, mistrust, and anger [21]. It has been shown that childhood trauma alters stress reactivity of the hypothalamus-pituitary-adrenal axis [22]. As adults, these individuals may develop physical, mental, and behavioral disorders with poor coping skills and feel very angry with themselves and society. Depression may result, and when individuals have other risk factors such as substance misuse, mistrust, and poor social support, they are at high risk for suicide.

Anger and Impulsive Behavior

Another well-known risk factor for suicide is anger and impulsiveness in individuals. These characteristics can play an important role in enhancing the risk of suicide [23, 24]. Teenagers and young adults are more impulsive, which may increase the risk of suicidal behavior [25]. This may be due to poor self-control and urgency to act on thoughts, disregarding implications and consequences.

Precipitating Factors

Mood Disorders

A strong association between suicide and depression has been documented [26]. The literature suggests that almost 90% of people who commit suicide have a mental disorder but only 5% of people with a mental disorder die by suicide [27]. Thus, depression in itself does not always lead to suicide. The stress-diathesis model describes the tendency to act upon suicidal ideations as based on multiple factors and not just the presence of a mental disorder such as depression [27]. Sometimes, unpleasant life events such as loss of a loved one or a relationship with extreme hopelessness may be so devastating that they lead to suicide [28].

Substance Misuse is an independent risk factor for suicide. Compared to the general population, people with alcohol and drug use disorders have a 10–14 times higher risk of suicide [29].

Chronic Illness

Patients with chronic illnesses, such as rheumatoid arthritis, diabetes, or cancer, may grieve for the loss of the life they experienced before being sick. Adjusting to the prolonged illness and effects, such as loss of employment, financial distress, or poor quality of life, is difficult. Such individuals may have protracted distress, develop depression and suicidal thoughts, and may even attempt suicide. Patients with chronic illness who have poor social support and quality of life are at high risk for suicide [30].

Challenges Faced by Military Veterans

The majority of first suicide attempts in veterans occur after separation from military service; veterans may find it difficult to adjust to civilian life due to relationship conflict, social isolation, pain from war injuries, and symptoms of post-traumatic stress disorder [31].

A 2019 report released by the Veterans Administration stated that the suicide rate among veterans is 1.5 times higher than among nonveterans [32, 33]. The literature indicates that major factors responsible for the high suicide rate in veterans are habituation and the acquired capability to attempt suicide [34]. Through repeated exposure to trauma or near-death experiences while serving on the front lines, soldiers may become habituated to trauma, injury, and death, and the fear of death may fade away. Another risk factor in this setting is survivors' guilt [35].

Financial Crisis/Loss of Job

Economic trends and industry shifts, such as corporate mergers and sale of companies in capitalistic economies or changing market forces, can cause major upheavals in individual lives through unemployment, loss of business, and failure to be able to support oneself. Many poor countries in Africa, South America, and Southern Asia are also known to have high suicide rates due to chronic hopelessness associated with the failure to earn enough income to raise a family or take care of the sick and elderly [36]. Again, the will to survive can vanish, propelling individuals in a highly distressed society to attempt suicide as a way to escape their financial conditions.

Environmental Factors

Environmental influences are well known to play a role in suicidal thoughts. Changes in human behavior triggered by negative external events can strongly affect the body's stress system and responses. These epigenetic changes influence psychiatric disorders and may lead to suicide attempts [37].

Perpetuating Factors

According to the stress-diathesis model, an ongoing biological or psychological factor in the absence of an acute event may not influence the individual toward a suicidal attempt. However, stressful acute events, such as divorce, rejection, or loss of a loved one, can cause a strong emotional experience of hopelessness and, when combined with impulsiveness, propel an individual toward suicide.

Suicide Risk Factors Specific to Cancer Patients

Although the literature documents a high risk of suicide in cancer patients, thus far, there is no clear documentation that cancer, in and of itself, is a risk factor for suicide in the absence of any psychopathology [38]. The risk of suicide in this patient population is high, however, since, in addition to general risk factors discussed above, cancer patients face other risk factors, including disease burden and life-changing adverse effects of cancer treatment. In the initial stage of disease, patients are anxious and fearful, with uncertainty of treatment outcome, and may experience treatment side effects, including nausea, poor energy, and sleep disturbances. As the disease progresses or treatment intensifies, patients may experience extreme fatigue, poor appetite, vomiting/diarrhea, neuropathy, and physical changes such as loss of muscle mass, loss of hair, and discoloration of skin and nail beds. Patients may lose their jobs and their physical independence. Depression may set in as the patient becomes more dependent on others. Role reversal within the family that comes with loss of independence, job loss, financial distress, and the potential guilt of being a burden on the family can intensify symptoms of depression.

In addition to hopelessness and fear of abandonment, one of the major risk factors for suicide in cancer patients is intolerable pain and suffering. Reports have documented that cancer patients do not always experience good pain control; for example, oncologists may fear prescribing opioids to emaciated or ill cancer patients with compromised liver, kidney, or respiratory functions.

Management of Suicidal Cancer Patients

Currently, the best screening tool is the clinical judgment of mental health providers and their expert opinions regarding suicide risk. Such providers working with these patients try to understand the root cause of these symptoms and provide support and management strategies. When a patient with cancer reports suicidal thoughts, it should be given serious consideration. It is important to determine whether such thoughts are due to physical challenges, such as unbearable pain, or because of hopelessness and depression. Many symptoms of depression, such as poor sleep, fatigue, and loss of appetite, are similar to those of cancer and the side effects of cancer treatment. Hopelessness, loss of meaning in life, despair, and suicidal thoughts are alarming signs [39]. Cancer patients with a history of psychopathology, previous suicide attempts, or a family history of mental illness are at high risk for suicide. Recognizing clinical depression in cancer patients is very important as depression is treatable; treatment may uplift patients' mood, giving them hope and motivating them to continue treatment.

Management of Suicidal Oncology Patients in the Emergency Department

About 8% of emergency department (ED) patients have suicidal thoughts, which they may not share unless asked [40]. It is important for the emergency physician to explore any suicidal thoughts from a patient with a history of mental illness. Based on the initial assessment, the clinician may need to consult the psychiatry service for a full evaluation and management. Management of suicidal oncology patients in the emergency department is a challenge due to time constraints, medical comorbidities needing immediate attention, the potential need to find availability of psychiatry beds in a med-psych facility, arranging transport to the appropriate hospital, and completing paperwork for involuntary admission, if the patient is not willing to go voluntarily.

The first step in management is to implement a proper safety protocol and protect the patient from self-harm. Patients with suicidal thoughts should not be allowed to leave the emergency department until they have had a complete evaluation by a mental health provider [41]. Medical staff should closely observe the patient, and the room should be searched for items that could be used as a weapon, such as a belt, wires, any sharp object, or glass medicine bottles [41].

Patients with vague suicidal thoughts but no plans, no previous attempts, no history of mental illness, no substance use, no agitation or confusion, and with good social support may be allowed to leave but with a firm follow-up plan [42], as the days after discharge from the emergency center are a high-risk period [40]. Patients who have any unbearable physical symptoms such as pain or severe nausea and vomiting should be admitted to the medical ward for stabilization and control of these symptoms.

Cancer patients with active suicidal thoughts who are undergoing active treatment or with medical comorbidities should be admitted to a hospital where they can be monitored closely until stabilized and their treatment continued. Daily assessment by a mental health provider, appropriate medication, and support are required. After discharge from the hospital, the patient should follow up with a psychiatrist for continued care.

It is important to determine if the suicidal thoughts are due to unbearable physical symptoms, such as severe pain or 569

intractable nausea/vomiting, which should be addressed and managed properly. Inadequate pain management in oncology is a multidimensional problem. It is a devastating symptom that many cancer patients suffer. Since the publication of the analgesic ladder in 1986, pain management has improved, but many cancer patients still do not receive proper pain control [43]. High pain levels, especially in advanced cancer stages, lead to suicidal thoughts. Pain symptoms should be properly managed by the palliative care team [7] during hospitalization, and patients should be discharged with adequate analgesics. Any other physical symptom should be properly investigated and managed.

Management of Suicidal Patients in the Oncology Clinic

Patients with active suicidal thoughts or plans to act on them require full evaluation by a mental health provider and should be admitted to a psychiatric facility for safety until stabilized. Patients with passive suicidal thoughts, with no plans and means to act on them, and with no major risk factors but good social support may be allowed to go home after thorough evaluation. With the patient's permission, a family member or a close friend can be involved who will monitor the patient's safety and adherence to treatment [42]. Such patients should be monitored through regular follow-up with a mental health provider and medication management as indicated.

Most suicidal patients do not want to end their lives; rather, they want to end their emotional pain and unbearable suffering. Talking about suicide does not trigger a suicidal attempt. Instead, it gives the patient a chance to discuss his/her fears and may help them find a source for sharing their feelings. Clinicians work in a collaborative model, walking with the patient on difficult roads to understand their emotional pain and suffering, validating their emotional turmoil, and helping them find alternative ways to cope. They assure these patients that the solution to their problems lies within them and they can work together to find those solutions [44]. They help raise patients' pain threshold and coping skills. Regular follow-up with a mental health provider for support and to learn and sustain coping strategies is key. The mental health provider develops a suicide-specific treatment plan and clinically tracks and documents ongoing suicidal risk. They also monitor medication compliance, side effects, and response. Most important is to empathize with the suicidal wish, identify protective factors, and work in a collaborative model.

The literature documents that combination therapy with medication and psychotherapy gives better results in managing depression [45]. This approach shortens the time until symptom control and increases adherence to medical treatment. Cognitive behavioral therapy and dialectal behavioral

therapy have shown good results to reduce suicidal thinking, as they address cognitive and behavioral aspects of suicide.

Antidepressants for the Treatment of Depression in Cancer Patients

Depression is the most common psychological symptom in cancer patients. It is often underdiagnosed or misdiagnosed because its symptoms overlap the physical symptoms of cancer and its treatment. The most frequent and common example of this misdiagnosis is hypoactive delirium. Prescribing an antidepressant or a stimulant to a delirious patient may exacerbate the condition [46]; hence, proper diagnosis and management of depression by a mental health provider is important. There are not enough data available to assess the efficacy and tolerability of antidepressants in the oncology setting, and no one antidepressant has been proven superior to another [47]. The most commonly prescribed antidepressants in the oncology setting are specific serotonin reuptake inhibitors (SSRIs). These drugs have a good safety profile with a minimum of drug interactions. The choice of antidepressant should be individualized to the specific symptoms in cancer patients. Patients with depression and poor sleep may be given antidepressants with sedative properties [48]. Depressed patients with pain symptoms may benefit from the choice of duloxetine. Cancer patients with nausea and poor appetite may benefit from mirtazapine; this drug not only helps with mood symptoms but also controls nausea and promotes appetite. It is important to understand that antidepressants take a few weeks to control symptoms. Psychostimulants deserve special consideration in terminally ill patients when a quick response is needed, as they provide a rapid effect [49]. Choosing an antidepressant becomes challenging when patients are on certain protocols with restrictions on the choice of antidepressant or when they are unable to swallow a pill. Communicating with the primary oncology team before starting an antidepressant, monitoring the side effects profile of the drug, and having a sound knowledge of drug-drug interactions is important.

Future Research

Patients with biopsychosocial risk factors and suicidal thoughts and behaviors may show a specific neural circuitry picture. Previous work on brain scans has identified specific circuits associated with executive functions, impulsivity, and mood [50]. Recently, it has been shown that every idea or thought has a specific pattern in the brain; these patterns can be identified through brain scans showing color changes in the brain neurons. Researchers have been working to identify suicidal thoughts through brain scans with the goal of

determining who could attempt suicide in the future. People with active suicidal thoughts exhibit increased brain activity in the areas associated with emotions and thoughts about self, which can be identified using functional MRI [51]. This work is still in the initial stages but seems promising. If thought processes reflecting risk factors of suicide can be identified and associated with interventions, a reduction in suicide attempts could result.

Conclusion

Suicide remains among the leading causes of death in the United States. The management of suicidal patients is a psychiatric emergency and the most challenging task faced by mental health providers. Through an empathic and collaborative approach, clinicians in the emergency department and mental health providers can prevent future injury and death. Emergency physicians, social workers, and mental health providers should work as a team to ensure the safety of suicidal patients, reduce emotional pain, address any acute medical condition leading to suicidal thoughts, and make appropriate treatment plans. Ensuring that treatment and safety plans are followed during transitions of care, when patients are transferred to another facility or referred for follow-up appointments, is essential to providing superior care and sustained stabilization.

Acknowledgment The author acknowledges David P Farris, MSIS, and Monica Modelska, who provided technical assistance for this chapter.

References

- World Health Organization. Suicide in the world: Global Health estimates. 9 Sep 2019. https://www.who.int/publications/i/item/ suicide-in-the-world. Accessed 27 Jul 2020.
- National Institute of Mental Health. Suicide. Last updated Sep 2020. https://www.nimh.nih.gov/health/statistics/suicide. shtml#part_155137. Accessed 28 Sep 2020.
- Zaorsky NG, Zhang Y, Tuanquin L, Bluethmann SM, Park HS, Chinchilli VM. Suicide among cancer patients. Nat Commun. 2019;10(1):207.
- Walker J, Waters RA, Murray G, Swanson H, Hibberd CJ, Rush RW, Storey DJ, Strong VA, Fallon MT, Wall LR, Sharpe M. Better off dead: suicidal thoughts in cancer patients. J Clin Oncol. 2008;26(29):4725–30.
- Chochinov HM, Wilson KG, Enns M, Mowchun N, Lander S, Levitt M, Clinch JJ. Desire for death in the terminally ill. Am J Psychiatry. 1995;152(8):1185–91.
- McFarland DC, Walsh L, Napolitano S, Morita J, Jaiswal R. Suicide in patients with cancer: identifying the risk factors. Oncology (Williston Park). 2019;33(6):221–6.
- Park SA, Chung SH, Lee Y. Factors associated with suicide risk in advanced cancer patients: a cross-sectional study. Asian Pac J Cancer Prev. 2016;17(11):4831–6.

- PDQ® Supportive and Palliative Care Editorial Board. PDQ Depression. Bethesda: National Cancer Institute. Updated 9 Jul 2019. Available at: https://www.cancer.gov/about-cancer/coping/feelings/depression-pdq. Accessed 28 Sep 2020. [PMID: 26389474].
- U.S. Department of Health & Human Services. Office for Civil Rights. HIPAA privacy rule and sharing information related to mental health. n.d. https://www.hhs.gov/sites/default/files/ hipaa-privacy-rule-and-sharing-info-related-to-mental-health.pdf. Accessed 10 Jul 2020.
- Meissner WW. Psychoanalytic notes on suicide. Int J Psychoanal Psychother. 1977;6:415–47.
- Chaney S. How 1930s psychoanalysts thought self-harm explained war. Psychol Today. 16 Jan 2018; https://www.psychologytoday.com/us/blog/the-history-self-harm/201801/how-1930spsychoanalysts-thought-self-harm-explained-war. Accessed 7 Jul 2020
- Sperber M. Suicide: Psychache and alienation. Psychiatric Times.
 Nov 2011. https://www.psychiatrictimes.com/view/suicidepsychache-and-alienation. Accessed 7 Jul 2020.
- Institute of Medicine (US) Committee on Pathophysiology and Prevention of Adolescent and Adult Suicide. Cognitive approaches to suicide. In: Suicide prevention and intervention: summary of a workshop. Washington: National Academies Press (US); 2001. https://www.ncbi.nlm.nih.gov/books/NBK223847/. Accessed 28 Sep 2020.
- Baumeister RF. Suicide as escape from self. Psychol Rev. 1990;97(1):90–113.
- Tavakoli HR. A closer evaluation of current methods in psychiatric assessments: a challenge for the biopsychosocial model. Psychiatry (Edgmont). 2009;6(2):25–30.
- Wright CD, Tiani AG, Billingsley AL, Steinman SA, Larkin KT, McNeil DW. A Framework for understanding the role of psychological processes in disease development, maintenance, and treatment: the 3P-Disease model. Front Psychol. 2019;10:2498.
- Yohn CN, Gergues MM, Samuels BA. The role of 5-HT receptors in depression. Mol Brain. 2017;10(1):28.
- Sharma S, Sharma J. Regulation of appetite: role of serotonin and hypothalamus. Iran J Pharmacol Therap. 2012;11(2):73–9.
- Menon V, Kattimani S. Suicide and serotonin: making sense of evidence. Indian J Psychol Med. 2015;37(3):377–8.
- Li H, Gao Z, Wu Q, Huang P, Lin C, Chen G. Relationship of hypothalamus-pituitary-adrenal (HPA) axis function and suicidal behavior in patients with depression. Shanghai Arch Psychiatry. 2013;25(1):32–9.
- Ports KA, Merrick MT, Stone DM, Wilkins NJ, Reed J, Ebin J, Ford DC. Adverse childhood experiences and suicide risk: toward comprehensive prevention. Am J Prev Med. 2017;53(3):400–3.
- Roy A. Gene–environment interaction and suicidal behavior. In: Dwivedi Y, editor. The neurobiological basis of suicide. Boca Raton: CRC Press/Taylor & Francis; 2012.
- Turecki G. Dissecting the suicide phenotype: the role of impulsive-aggressive behaviours. J Psychiatry Neurosci. 2005;30(6):398–408.
- Gvion Y, Levi-Belz Y, Hadlaczky G, Apter A. On the role of impulsivity and decision-making in suicidal behavior. World J Psychiatry. 2015;5(3):255–9.
- 25. Kasen S, Cohen P, Chen H. Developmental course of impulsivity and capability from age 10 to age 25 as related to trajectory of suicide attempt in a community cohort. Suicide Life Threat Behav. 2011;41(2):180–92.
- American Association of Suicidology. Some facts about suicide and depression. 23 Jun 2009. http://www.dartmouth.edu/eap/library/ depressionandsuicide.pdf. Accessed 7 Jul 2020.
- Brådvik L. Suicide risk and mental disorders. Int J Environ Res Public Health. 2018;15(9):2028.

- Yen S, Pagano ME, Shea MT, Grilo CM, Gunderson JG, Skodol AE, et al. Recent life events preceding suicide attempts in a personality disorder sample: findings from the collaborative longitudinal personality disorders study. J Consult Clin Psychol. 2005;73(1):99–105.
- 29. Esang M, Ahmed S. A closer look at substance use and suicide. Am J Psychiatry Residents' J. 2018;13(6):6–8.
- Turner J, Kelly B. Emotional dimensions of chronic disease. West J Med. 2000;172(2):124–8.
- Villatte JL, O'Connor SS, Leitner R, Kerbrat AH, Johnson LL, Gutierrez PM. Suicide attempt characteristics among veterans and active-duty service members receiving mental health services: a pooled data analysis. Mil Behav Health. 2015;3(4):316–27.
- 32. Wikipedia. United States military veteran suicide. Last edited 23 Aug 2020 at 18:23 https://en.wikipedia.org/wiki/United_States_ military_veteran_suicide. Accessed 28 Sep 2020.
- 33. U.S. Department of Veterans Affairs. Office of Mental Health and Suicide Prevention. National Veteran Suicide Prevention Annual Report. 2019 https://www.mentalhealth.va.gov/docs/datasheets/2019/2019_National_Veteran_Suicide_Prevention_Annual_ Report_508.pdf. Accessed 28 Sep 2020.
- 34. Selby EA, Anestis MD, Bender TW, Ribeiro JD, Nock MK, Rudd MD, et al. Overcoming the fear of lethal injury: evaluating suicidal behavior in the military through the lens of the Interpersonal-Psychological Theory of Suicide. Clin Psychol Rev. 2010;30(3):298–307.
- Opp RE, Samson AY. Taxonomy of guilt for combat veterans. Prof Psychol Res Pr. 1989;20(3):159–65.
- 36. Bantjes J, Iemmi V, Coast E, Channer K, Leone T, McDaid D, et al. Poverty and suicide research in low- and middle-income countries: systematic mapping of literature published in English and a proposed research agenda. Glob Ment Health (Camb). 2016; 3:e32.
- Nauert R. Environmental factors may trigger suicide genes. Psych Central. 8 Aug 2018 https://psychcentral.com/news/2018/01/03/ environmental-factors-may-trigger-suicide-genes/130775.html. Accessed 6 Jul 2020.
- Filiberti A, Ripamonti C, Totis A, Ventafridda V, De Conno F, Contiero P, Tamburini M. Characteristics of terminal cancer patients who committed suicide during a home palliative care program. J Pain Symptom Manag. 2001;22(1):544–53.
- Breitbart W. Cancer, depression, and suicide risk: signs to watch for. Cancer.Net. 1 Aug 2018. https://www.cancer.net/blog/2018-08/ cancer-depression-and-suicide-risk-signs-watch. Accessed July 6, 2020.
- Betz ME, Boudreaux ED. Managing suicidal patients in the emergency department. Ann Emerg Med. 2016;67(2):276–82.
- Carrigan CG, Lynch DJ. Managing suicide attempts: guidelines for the primary care physician. Prim Care Companion J Clin Psychiatry. 2003;5(4):169–74.
- Norris D, Clark MS. Evaluation and treatment of the suicidal patient. Am Fam Physician. 2012;85(6):602–5.
- Carlson CL. Effectiveness of the World Health Organization cancer pain relief guidelines: an integrative review. J Pain Res. 2016;9:515–34.
- 44. Jobes DA. Managing suicidal risk: a collaborative approach. 2nd ed. New York: The Guilford Press; 2016.
- 45. Dunlop BW. Evidence-based applications of combination psychotherapy and pharmacotherapy for depression. Focus. 2016;14(2):156–73.
- 46. Rosenstein DL. Depression and end-of-life care for patients with cancer. Dialogues Clin Neurosci. 2011;13(1):101–8.
- 47. Ostuzzi G, Matcham F, Dauchy S, Barbui C, Hotopf M. Antidepressants for the treatment of depression in people with cancer. Cochrane Database Syst Rev. 2018;4(4): CD011006.

- 48. Ahmed E. Antidepressants in patients with advanced cancer: when they're warranted and how to choose therapy. Oncology (Williston Park). 2019;33(2):62–8.
- 49. Block SD. Assessing and managing depression in the terminally ill patient. ACP-ASIM End-of-Life Care Consensus Panel. American College of Physicians-American Society of Internal Medicine. Ann Intern Med. 2000;132(3):209–18.
- 50. Pattani A. Researchers look to brain images to predict who will attempt suicide. Philadelphia Inquirer. 12 Sep 2018. https://medi-

calxpress.com/news/2018-09-brain-images-suicide.html. Accessed 28 Sep 2018.

 University of Illinois at Chicago. Brain scans may provide clues to suicide risk. 10 Oct 2019. Science Daily. 10 Oct 2019. https://www. sciencedaily.com/releases/2019/10/191010113244.htm. Accessed 7 Jul 2020.

Substance Abuse



45

Adam Rzetelny, Tatiana Starr Kreitzer, Maria A. Farberov, and Steven D. Passik

Introduction

Until very recently, discussing the problem of substance abuse in people with cancer might have been seen as having only minor clinical relevance. Certainly, the relief of anxiety, pain, and other distressing symptoms has been identified as a top priority for oncologic clinicians, and the use of medications, even those with abuse potential, has been deemed essential. Much of the literature on this subject suggests that problems of substance abuse are only infrequently encountered in oncologic. Perhaps this underestimation of the problem stems from the fact that much of this academic work has come from tertiary care settings-where those with histories of addiction are less frequently encountered because of barriers to care in the form of economic issues, lack of insurance coverage, and estrangement from healthcare providers who might diagnose and refer patients to such centers. Or perhaps it has been that cancer typically remains a disease of the sixth decade of life and beyond, whereas addiction overwhelmingly manifests earlier in life, making it unlikely to emerge de novo in a person first exposed to substances with abuse potential when they are older and ill [1, 2]. Or perhaps it has been because cancer used to follow an almost uniformly fast and fatal trajectory, and so any exposure to controlled substances was likely to be brief and occur during a period of time when the person was becoming increasingly disabled and less likely to engage in practices related to the obtaining

A. Rzetelny (🖂)

Department of Medical Affairs, Collegium Pharmaceuticals, Stoughton, MA, USA

e-mail: arzetelny@collegiumpharma.com

T. S. Kreitzer

University of South Florida College of Nursing, Tampa, FL, USA

M. A. Farberov

S. D. Passik

Scientific Affairs, Research and Development, Collegium Pharmaceutical, Stoughton, MA, USA and use of illicit drugs. Thus, even if the exposure to such drugs might trigger a relapse in a person who suffered with the disease of addiction before they became ill with cancer, the dysfunctional behaviors that might have been set in motion would be mediated and limited by the relentless impact of the cancer itself. Or perhaps it was simply the trivialization of addiction that characterized the early rhetoric accompanying the increase in opioid prescribing that led to this being such a neglected topic [3].

In response to the public health crisis that is chronic pain in our aging society, the prescribing of opioids and other controlled substances increased dramatically. Unfortunately, a parallel set of public health crises arose: the problems of prescription drug abuse, diversion, overdose, and death. Although a substantial portion of the problem of opioidrelated overdoses now appears to be fueled by the prevalence of illicitly manufactured fentanyl, prescription opioids remain a central concern [4]. Given that people with cancer are living longer at all stages of disease, including those with painful but stable disease and those who go onto remission but are left with chronic pain issues from chemotherapy and other factors, exposures to controlled substances are considerably longer than they once were, and thus there is greater opportunity for those who come to the disease with a history of substance use disorder (SUD) to lose control and overuse or even have the problem of addiction fully rekindled [5, 6]. Thus, it is not surprising that there has been a substantial increase in the number of opioid-related emergency department (ED) visits from patients with cancer [7], with nearly a fifth of cancer patients in the ED admitting to past or current misuse and a third at high risk of misusing their opioid prescriptions [8].

The recently exacerbated problems of opioid abuse and overdose also affect patients who do not themselves suffer from these problems. A growing concern for those prescribing controlled substances and treating pain, anxiety, and other symptoms in people with cancer, including older patients, is that their medications are increasingly sought after by younger individuals with substance abuse problems

Clinical Research Education, Department of Medicine, Northwell Health, Staten Island University Hospital, Staten Island, NY, USA

K. H. Todd et al. (eds.), Oncologic Emergency Medicine, https://doi.org/10.1007/978-3-030-67123-5_45

in their family or social environment (including grandchildren and caregivers). Additionally, recent data suggests that some parents of children with cancer do not manage their children's opioids safely, at times giving them to another person or even taking the opioid themselves [9]. Thus, it is not surprising that in the midst of the current opioid crisis, new restrictions and limits have emerged on the prescribing of opioids, including in the ED [10]. Although the intended aim of these restrictions is to address the management of acute, non-cancer pain, cancer patients in the ED and elsewhere may be impacted. An unintended consequence of the recent well-intended restrictions has been to increase the stigmatization of patients in pain, with deleterious effects on healthcare outcomes [11, 12]. Cancer patients are not immune to this burden, as evident in the lack of adequate pain control experienced by some opioid-tolerant cancer patients in the ED [13]. Mitigating the impact of stigma on pain and opioids requires a better understanding of addiction.

Thus, it is not only important for oncologic professionals of every stripe to have a working knowledge of addiction medicine principles and practices but particularly the oncologic ED professional. In the ED, consequences of drug abuse—from the older person with cancer presenting in withdrawal because family members have stolen their medication to the younger person whose addiction has been rekindled by the need for pain medication—are likely to be common. Common doesn't mean obvious, however, and it is important to dispel myths about the relative absence of addiction issues in cancer to help emergency care providers anticipate such problems and learn to manage them.

The increasing use of opioids in non-cancer pain grew out of recognition that people with cancer pain (at least those seeking treatment at tertiary care facilities) appeared to be able to take these medications with generally positive results. That is, their pain was controlled, side effects manageable, functional status improved or stabilized, and problems of misuse or addiction minimal. Opioid prescribing then increased dramatically, particularly in North America, to the much more diverse population of those with chronic painmore diverse in terms of age, psychiatric, and addiction histories and comorbidities, as well as in duration of exposure [14]. Not surprisingly, the results of this effort were mixed. Cancer pain management with opioids follows a basically self-titration model consistent with an assumption that risk of misuse and addiction is uniformly minimal across patients. When this method of delivering opioid therapy in non-cancer pain began to meet with problems of abuse and diversion, a risk stratification model began to emerge. Younger age, personal or family history of addiction, history of sexual trauma, and active psychiatric comorbidity were seen as risks for a poor outcome in opioid therapy, unless the delivery of this therapy was tailored to the needs of the individual (with the employment of safeguards such as urine drug testing, prescription monitoring programs, and the like, as well as consultations with psychiatric and addiction professionals to assure safety). There is a certain irony in the fact that oncologic pain management must now take a page from the noncancer pain "playbook." This type of risk stratification is somewhat foreign to oncologic pain management, but it seems that the time is right to close the loop and for the therapy that initially influenced non-cancer pain practice to adopt strategies developed therein, especially now that many of the differences between cancer and non-cancer pain patients have narrowed [15].

Pain and anxiety management are not the only aspects of cancer care affected by the presence of a SUD. Indeed, because unchecked drug or alcohol abuse can cause spotty or complete nonadherence to potentially lifesaving cancer treatments, virtually every step along the disease trajectory, from diagnosis to palliative care, can be threatened. Thus, the "downstream" complications of substance abuse can lead to a person with cancer presenting in the ED with problems related to nonadherence of every variety. A question is whether the ED professional will recognize them as such. And if the ED professional is working outside of a tertiary care academic center, the frequency with which they will be confronted with SUDs is shockingly high, due to the high base rate of these disorders in this population which is so much more reflective of the population as a whole. Particularly when one considers that substance use can be a risk factor for cancer, one would expect substance abusers to be over- and not underrepresented in the oncologic population. Many oncologic ED professionals from nearly all of the disciplines represented in this group of practitioners are lacking in their knowledge of addictions. There is an enormous gap between the prevalence of these problems and the expertise available to care for cancer patients who are struggling with them. It is hoped that this chapter helps bridge this gap for oncologic emergency medicine.

Scope of the Problem

Substance Use Disorder

Substance use disorders are a consistent phenomenon in the USA over time, with estimated base rates of 6–15% [16–20]. This prevalence of drug abuse certainly touches medically ill patients and can negatively influence how patients are treated. Although few studies have been conducted to evaluate the epidemiology of substance abuse in patients with advanced illness, these disorders have been reported to be relatively rare within the tertiary care population with cancer and other advanced diseases [21–23]. However, the prevalence of alcoholism in major cancer centers is most likely underestimated. A study by Bruera and colleagues [24] of

100 terminally ill alcoholic cancer patients found that despite multiple hospital admissions and screenings, only one-third had documentation of alcoholism in their medical records. How then would the ED professional know to anticipate and plan for emergent problems related to alcohol abuse in a person presenting in their setting?

The belief that drug habits diminish and vanish with age is no longer held with the certainty of past belief. An early study supporting this belief reported that 50% of individuals addicted to narcotics were no longer active drug users by age 32 and over 99% were no longer users by age 67 [25]. However, as the "baby boom" cohort ages, the extent of alcohol and medication misuse is predicted to increase significantly because of the combined effect of the growing population of older adults and cohort-related differences in lifestyles and attitudes [26]. One study suggested that the number of illicit drug users aged 50 years or older will double from the year 2000 to the year 2020 because of an anticipated 52% increase in this segment of the population and the attendant shift in attitudes and historical experiences with substance use in this cohort [27].

Prescription Drug Abuse

The use of illicit drugs and the nonmedical use of prescription opioids have increased significantly in the general population over the last decade, though it has begun to decline in the just the past few years, and the highest prevalence remains among younger adult men [16, 28]. Such estimates, however, belie alarming trends emerging among older adults. Among adults aged 50 or older, nearly five million, roughly 5% of that age group, report using illicit drugs in the past year [20]. Marijuana is the most abused drug in the USA, but among adults aged 60 or older, the abuse of prescription drugs is equally common. A changing pattern of cannabis use among older adults suggests that as an individual ages, the social incentive to smoke marijuana decreases, while the attempt to use it medicinally increases [29]. In the oncologic setting, this might include an attempt to self-medicate for nausea, anorexia, pain, and anxiety or combinations of these common symptoms [30]. More alarming than rates of cannabis use, ED visits related to pharmaceutical abuse more than doubled from 2004 to 2008 among adults aged 50 or older, and a fifth of these was among adults aged 70 or older [31]. Prescription opioids were the most common culprit, followed by benzodiazepines. Although ED visits in 2008 related to illicit drug use among adults 50 and older were a little less than half that of pharmaceuticals (118,495 vs. 256,097 visits), they were not uncommon [31, 32]. The majority of those visits were related to cocaine, followed by heroin. Consistent with this, one study demonstrated marijuana, cocaine, and opioid use in 2.4%, 1.9%, and 11.6% of elderly men, respectively [33]. Substance use treatment admissions among adults aged 50 and older have nearly doubled in recent years, from 6.6% of all admissions in 1992 to 12.7% in 2009 [34]. During this same time period, alcohol as the only substance of abuse being treated decreased from 87.6% to 58.0%, while the addition of other drugs to alcohol increased from 12.4% to 42.0%. Also around this time, treatment admissions involving heroin more than doubled, from 7.2% to 16.0%, and those reporting multiple drugs of abuse nearly tripled [35].

Even as the baby boomer population ages and more frequently experiences pain, there is a paucity of information on older patients and the risk of comorbid pain and SUDs. A survey in Denmark revealed that 22.5% of men and 27.8% of women aged 65 and older reported chronic pain [36]. Of these men and women, 35% of them were not satisfied with their pain treatment. This can lead to alternative methods for relieving pain such as taking non-prescribed medications. In one study of 100 patients with chronic pain (average age near 50), 23 tested positive for illicit drugs, and 12 tested positive for opioids, even though they had no prescription and denied taking opioids [37]. In another study of primary care patients in a Veterans Affairs facility who were receiving opioids for the treatment of chronic pain (average age 59), 78% reported at least one indicator of medication misuse during the prior year, with significantly more of those who misused pain medications reporting comorbid SUD [38]. This is consistent with a more recent examination of a subset of data from the Researched Abuse, Diversion and Addiction-Related Surveillance (RADARS) system's finding that though severe chronic pain is common in adults entering treatment for prescription opioid abuse, it is exponentially more prevalent in adults older than 45 years (70%) relative to the 18-24-year age group (45%) [1]. Clearly, to the extent that chronic pain and SUDs are comorbid or mutually exacerbating problems, older adults are a particularly vulnerable population. This might be especially true in the oncologic culture, wherein performing a risk assessment has been historically uncommon and decreasing patients' wariness about using opioids aggressively when needed has been the biggest concern.

Thus, the emerging pattern, consistent with the aging of the "baby boom" generation and their greater likelihood of exposure to various types of drugs, is that illicit and prescription drug misuse and abuse, along with the need for treatment, is expected to double by 2020 (relative to the 1990s prevalence estimates) among older adults [27, 39], with the greatest changes reflecting the increasing rates of ED visits and treatment admissions related to prescription opioids, benzodiazepines, heroin, and cocaine. Knowledge of these trends should assist oncologic providers in identifying and managing problems in a more age-appropriate manner.

Alcohol

There have been relatively few studies examining the prevalence of alcoholism in an oncologic population. The prevalence likely varies widely from one cancer to another with the highest rate in the head and neck cancer population. One study found that greater than 25% of patients admitted to a palliative care unit had problems with alcohol abuse [24]. Socioeconomic barriers such as low income or unemployment, lack of health insurance, and possibly even attempts to self-medicate early symptoms of cancer may preclude patients from seeking care at tertiary care centers. Furthermore, alcohol abuse complicates cancer care. For example, postsurgical withdrawal and delirium tremens (DTs) can be life-threatening. Unfortunately, many patients are unrecognized prior to undergoing surgery. Integrating screening for alcoholism and offering detoxification ahead of surgery are underutilized opportunities, discussed in detail later in this chapter.

Diagnostic and Treatment Issues

Defining Substance Use Disorder and Addiction in the Medically III

It is difficult to define substance use disorder (SUD) and addiction in patients with cancer, as the definitions of terms have been adopted from populations that do not necessarily have a medical illness. Furthermore, the pharmacological phenomena of tolerance and withdrawal may be present in SUD but are distinct from it, and it is important that clinicians understand the difference. To complicate the matter further, the use of these terms is inconsistent across settings and is so influenced by sociocultural considerations that it may lead to confusion in the clinical setting. In some settings, the terms "abuse" and "addiction" have been shunned in favor of the less stigmatizing "substance use disorder." (In this chapter, we use these terms somewhat interchangeably.) What these concepts have in common is some degree of the use of illicit drugs or the non-medical use of prescription medications, and at the most severe end of the spectrum there is craving and continued or compulsive use despite harms. Harms may be social, physical, psychological, or vocational in nature. The severity of the SUD depends, in part, on the harms and whether those harms are actual or potential. For example, a patient who continues to smoke following a lung cancer diagnosis likely has a more severe addiction than a smoker without cancer, even though they may smoke the same daily amount. This distinction is clinically relevant because understanding the severity helps determine the needed level of medical response. A thorough review of this *Physical Dependence*: An expected biological adaptation to a drug that results in tolerance and withdrawal. Physical dependence is distinct from SUD or addiction which may involve craving and continued or compulsive use despite harms.

- "Tolerance" occurs when a higher dosage of a drug is required over time to achieve the same effect.
- "Withdrawal" occurs when rapid discontinuation of a medication leads to a sequalae of adverse symptoms.

Because substance use is increasingly widespread in the population at large, patients with cancer who have used illicit drugs are more frequently encountered in medical settings. Illicit drug use, actual or suspected misuse of prescribed medication, or actual SUDs create the most serious difficulties in the clinical setting, complicating the treatment of pain. However, the management of SUD is fundamental to medical therapy adherence and safety during treatment. Also, adverse interactions between illicit drugs and medications prescribed as part of the patient's treatment can be dangerous. The presence of a SUD may alienate or weaken an already tenuous social support network that is crucial for alleviating the chronic stressors associated with advanced disease and its treatment. Therefore, a history of substance use can impede treatment and pain management and increase the risk of hastening morbidity and mortality among those with advanced cancer, which can only be alleviated by a therapeutic approach that addresses drug-taking behavior while expediting the treatment of the malignancy and distressing symptoms, as well as addiction [21, 22].

Important factors when assessing drug-taking behavior in cancer patients:

- Undertreatment of associated issues, particularly pain disorders.
- Sociocultural differences in defining "aberrant" drug taking.

Varied and repeated observations over a period of time may be necessary to categorize questionable behaviors properly (Table 45.1). Perceptive psychiatric assessment is crucial and may require evaluation by consultants who can elucidate the complex interactions among personality factors and psychiatric illness.

Patients with borderline personality disorders, for example, may impulsively use prescription medications that regulate inner tension or improve chronic emptiness or boredom

Possible alternate diagnoses for aberrant drug-taking behaviors				
Anxiety				
Depression				
Insomnia				
Problems of adjustment (such as boredom caused by decreased				
ability to engage in usual activities)				
Borderline personality disorder				

 Table 45.1
 Differential diagnoses to consider when interpreting aberrant drug-taking behaviors

and express anger at physicians, friends, or family. Psychiatric assessment is vitally important for both the population without a prior history of substance abuse and the population of known substance abusers who have a high incidence of psychiatric comorbidity [51].

Pseudoaddiction

Various studies have provided compelling evidence that pain is poorly treated in many oncologic patients [42–44]. Clinical experience indicates that the inadequate management of symptoms and related pain may be the motivation for aberrant drug-taking behaviors.

Pseudoaddiction: distress and aberrant drug-seeking behaviors that produce a similar pattern as individuals with substance use disorders; however, these behaviors actually stem from the patient seeking relief from untreated pain [45]:

- Patients are often attempting to "self-medicate," and behaviors can be considered pseudoaddictive if sufficient pain relief eliminates these behaviors.
- Physical dependence can often lead to pseudoaddictive behaviors, as clinicians do not compensate for growing tolerance to medications and therefore underdose patients.

More recent scientific advances have also provided new insight into behaviors that may be considered pseudoaddiction. Pharmacogenetic variances in the enzymes that metabolize pain medications help to explain individual differences in medication response and side effects experienced. If a patient is an ultrarapid metabolizer, they may complain that the medication is effective for a shorter period of time than is common for that medication. If a patient is a poor metabolizer, they may complain that the medication is not working or possibly continue to ask for increased amount of medication. Pharmacogenetic variations should be considered and pharmacogenetic testing implemented when a patient has an unusual response to a medication, more than expected side effect profile, and/or inefficacy at usual dosages [46].

The potential for pseudoaddiction creates a challenge for the assessment of a patient with both a substance use disorTable 45.2 Examples of aberrant drug-taking behaviors and severity

Examples of clearly aberrant behaviors	Examples of potentially aberrant behaviors
Illicit drug use	Requests for early medication refills
Intravenous injection of oral formulations	Requesting specific medications
Recurrent prescription "losses"	Patient taking extra doses of medication

der and an advanced illness. Clinical evidence indicates that aberrant behaviors impelled by unrelieved pain can become so dramatic in this population that some patients appear to return to illicit drug use as a means of self-medication. Others use more covert patterns of behavior, which may also cause concerns regarding the possibility of true addiction.

Aberrant behaviors set in motion by poorly relieved pain can, in vulnerable patients, lead to actual loss of control of their opioids and/or other controlled substances, and so pseudoaddiction and abuse or addiction should not be considered mutually exclusive. Pseudoaddiction might be best thought of as akin to the problem of secondary alcoholism in response, for example, to an untreated panic disorder. If selfmedication of panic disorder is persistent for some duration. before long a patient will likely have both problems-panic disorder and alcoholism. People taking liberties with their opioid dosing should be considered for a parallel diagnosis of addiction if attempts to improve analgesia do not lead to a relatively rapid cessation of the problematic behaviors [45]. Although it may not be obvious that drug-related behaviors are aberrant, the meaning of these behaviors may be difficult to discern in the context of unrelieved symptoms [21-23]. This can be a particularly vexing issue when the person with cancer presents in the ED. Is the presentation related to poor pain control or substance abuse, or both?

Aberrant Drug-Taking Behaviors

When a drug is prescribed for a medically diagnosed purpose, less assuredness exists as to the behaviors that could be deemed aberrant, thereby increasing the potential for a diagnosis of drug abuse or addiction. The ability to categorize these questionable behaviors as apart from social or cultural norms is also based on the assumption that certain parameters of normative behavior exist. Although it is useful to consider the degree of aberrancy of a given behavior, it is important to recognize that these behaviors exist along a continuum, with certain behaviors being less aberrant (such as aggressively requesting medication) and other behaviors more aberrant (such as injection of oral formulations) (Table 45.2). If a large portion of patients were found to engage in a certain behavior, it may be normative, and judgments regarding aberrancy should be influenced accordingly [21–23].

We know more scientifically about aberrant behaviors, their prevalence, and meaning today than we did in the mid-1990s. We know that many patients will have at least a few aberrant behaviors in a 6-month period [47]. We also know that once a patient has demonstrated four behaviors in their lifetime, they have an 85% likelihood of meeting diagnostic criteria for substance use disorder [48]. But there is still much to be learned, confirmed, replicated, and studied.

Disease-Related Variables

Changes caused by progressive diseases, such as cancer, also challenge the principal concepts used to define addiction. Alterations in physical and psychosocial functioning caused by advanced illness and its treatment may be difficult to distinguish from the morbidity associated with drug abuse. In particular, alterations in functioning may complicate the ability to evaluate a concept that is vital to the diagnosis of addiction: "use despite harm." For example, discerning the questionable behaviors can be difficult in a patient who develops social withdrawal or cognitive changes after brain irradiation for metastases. Even if diminished cognition is clearly related to pain medication used in treatment, this effect might only reflect a narrow therapeutic window rather than the patient's use of analgesic to acquire these psychic effects [40, 41, 48]. To accurately assess drug-related behaviors in patients with advanced disease, explicit information is usually required regarding the role of the drug in the patient's life. Therefore, the presence of mild mental clouding or the time spent out of bed may have less meaning than other outcomes, such as noncompliance with primary therapy related to drug use or behaviors that threaten relationships with physicians, other healthcare professionals, and family members [40, 41, 49].

Risks in Patients with Current or Remote Histories of Drug Abuse

There is a lack of information regarding the risk of abuse or addiction during or subsequent to the therapeutic administration of potentially abusable drugs to medically ill patients with a current or remote history of abuse or addiction [40, 41]. The possibility of successful long-term opioid therapy in patients with cancer or chronic nonmalignant pain has been indicated by anecdotal reports, particularly if the abuse or addiction is remote [52–54].

Because it is commonly accepted that the likelihood of aberrant drug-related behavior occurring during treatment for medical illness will be greater for those with a remote or current history of substance abuse, it is reasonable to consider the possibility of abuse behaviors occurring when using different therapies. For example, although no clinical evidence exists to support the notion that the use of short-acting drugs or the parenteral route is more likely to cause questionable drug-related behaviors than other therapeutic strategies, it may be prudent to avoid such therapies in patients with histories of drug abuse [40, 41].

Clinical Management of Substance use Disorders in Oncologic

The greatest challenges in caring for patients with SUDs and cancer typically arise when these patients are actively abusing alcohol or other drugs. This is in part because they experience more difficulty in managing pain [55]. Patients may become caught in a cycle where pain functions as a barrier to seeking treatment for addiction possibly complicating treatment for chronic pain [56]. Also, because pain is undertreated, the risk of bingeing with prescription medications and/or other substances increases for drug-abusing patients [55]. The implementation of a more comprehensive treatment plan for such patients may indeed only be initiated after ED visit(s) bring the need for such a labor-intensive program to light.

General Recommendations

The following guidelines can be beneficial, whether the patient is actively abusing drugs or has a history of substance abuse. The principles outlined assist clinicians in establishing structure, control, and monitoring of addiction-related behaviors, which may be helpful and necessary at times in all pain treatment [21].

Recommendations for the long-term administration of potentially abusable drugs, such as opioids, to patients with a history of substance abuse are based exclusively on clinical experience. Research is needed to ascertain the most effective strategies and to empirically identify patient subgroups which may be most responsive to different approaches. The following guidelines broadly reflect the types of interventions that might be considered in this clinical context [37, 49, 57].

Multidisciplinary Approach

Pain and symptom management is often complicated by various medical, psychosocial, and administrative issues in the population of advanced patients with a substance use disorder. The most effective team may include a physician with expertise in pain/palliative care, nurses, social workers, and, when possible, a mental healthcare provider with expertise in the area of addiction medicine [21, 22].

Assessment of Substance Use History

In an effort to not offend, threaten, or anger patients, clinicians frequently avoid asking patients about drug abuse. There is also often the expectation that patients will not answer truthfully. However, obtaining a detailed history of duration, frequency, and desired effect of drug use is vital. Adopting a nonjudgmental position and communicating in an empathetic and truthful manner is the best strategy when taking patients' substance abuse histories [21–23]. Avoid stigmatizing terms such as "abuse." Do not rush through a substance use assessment: It helps to make eye contact and show the patient that you really care about their answers. Patients can be quite candid about their drug and alcohol use, though even then it is not unusual for them to underreport or minimize their problems.

In anticipating defensiveness on part of the patient, it can be helpful for clinicians to mention that patients often misrepresent their drug use for logical reasons, such as stigmatization, mistrust of the interviewer, or concerns regarding fears of undertreatment. It is also wise for clinicians to explain that in an effort to keep the patient as comfortable as possible, by preventing withdrawal states and prescribing sufficient medication for pain and symptom control, an accurate account of drug use is necessary [21–23].

Taking an accurate, detailed history from the patient is essential for the proper assessment and treatment of alcohol and drug abuse as well as any comorbid psychiatric disorders. It is also important to ask about the duration, frequency, and desired effect of drug or alcohol consumption. In the wake of current pressures to treat the majority of patients in ambulatory settings and to admit patients on the morning of major surgery, the quick identification of alcoholism and initiation of plans for social, medical, and physiological needs of the patient must begin upon initial contact.

The use of a careful, graduated-style interview can be beneficial in slowly introducing the assessment of drug abuse. This approach begins with broad and general inquiries regarding the role of drugs in the patient's life, such as caffeine and nicotine, and gradually proceeds to more specific questions regarding illicit drugs. This interview style can also assist in discerning any coexisting psychiatric disorders, which can significantly contribute to aberrant drug-taking behavior. Once identified, treatment of comorbid psychiatric disorders can greatly enhance management strategies and decrease the risk of relapse [21–23].

Use of Risk Assessment Tools

As stated above, potential opioid use must be accompanied by risk stratification and management. Given time constraints, a full psychiatric interview may not be feasible, and thus time-sensitive measures are clearly needed to help in this endeavor. Many screening tools contain items on personal and family history of addiction as well as other historyrelated risk factors, such as preadolescent sexual abuse, age, and psychological disease. These are tools for clinical decision-making and should not be viewed as necessarily diagnostically accurate [58, 59]. Whatever tool the clinician chooses, it is advised that the screening process be presented to the patient with the assurance that no answers will negatively influence effective treatment. One risk factor unique to the oncologic setting is the economic pressure that accompanies the disease and its treatment. The depletion of savings over time can be a huge stress, and for some the temptation to divert medications with a street value may be seen as a matter of survival.

Setting Realistic Goals for Therapy

The rate of recurrence for drug abuse and addiction is high in general. The stress associated with cancer and the easy availability of centrally acting drugs increase this risk. Therefore, total prevention of relapse may be impossible in this type of setting. Gaining an understanding that compliance and abstinence are not realistic goals may decrease conflicts with staff members in terms of management goals. Instead, the goals might be perceived as the creation of a structure for therapy that includes ample social/emotional support and limit setting to control the harm done by relapse [21–23].

There may be some subgroups of patients who are unable to comply with the requirements of therapy because of severe substance use disorders and comorbid psychiatric diagnoses. In these instances, clinicians must modify limits on various occasions and endeavor to develop a greater variety and intensity of supports. This may necessitate frequent team meetings and consultations with other clinicians; however, pertinent expectations must be clarified, and therapy that is not successful should be modified [21–23].

Evaluation and Treatment of Comorbid Psychiatric Disorders

Extremely high comorbidities of personality disorders, depression, and anxiety disorders exist in alcoholics and other patients with substance abuse histories [51]. Individuals with a history of alcohol abuse have been found to be at higher risk for other psychiatric disorders [60]. The most common comorbid mental disorders associated with alcoholism are anxiety disorders (19.4%), antisocial personality disorder (14.3%), affective disorder (13.4%), and schizophrenia (3.8%) [61]. The occurrence of comorbid mental disorders in alcoholics may contribute to poor treatment compliance and success due to cognitive limitations and premorbid (in relation to the diagnosis of cancer) pain and neurological deficits. The same is true of opioid abuse where 85% of addicts have a comorbid, nondrug abuse-related psychological disorder [51]. Thus, the ED professional assessing the cancer patient with addiction or alcoholism must anticipate and identify for treatment or referral any comorbid disorders present. The treatment of depression and anxiety can increase patient comfort and decrease the risk of relapse or aberrant drug taking [21–23].

Preventing and Minimizing Withdrawal Symptoms

Assessing and treating withdrawal can help mitigate the immediate harms, as well as preventing substance use relapse. Identifying withdrawal, as with uncovering substance misuse, is an opportunity to initiate a useful discussion with the patient. Because patients with drug abuse histories often use multiple drugs, it is necessary to conduct a complete drug-use history to prepare for the possibility of withdrawal. A thorough discussion of this topic is beyond the scope of this chapter, but we will briefly discuss opioid and alcohol withdrawal, as they can be particularly consequential and likely to be encountered in the ED. Opioid withdrawal can be extremely unpleasant but rarely results in death. However, for a patient on long-term opioid therapy for pain, withdrawal can precipitate a severe exacerbation of pain, and in an individual with an opioid use disorder (OUD), withdrawal also increases the risk of a relapse, which can sometimes be fatal [62]. Perhaps more commonly, withdrawal (and indeed sometimes the ED visit itself) can indicate that the patient may have increased their use of prescription opioids, running out early. In this case, an important question is whether there has been an exacerbation of pain secondary to a change in the disease progression. Accurately assessing withdrawal can be challenging, as the symptoms associated with different types of withdrawal can overlap. To aid in

assessing withdrawal, it is helpful to utilize the substance use history, conducted in a nonjudgmental manner so as not to evoke shame and silence. Simply noticing some possible symptoms of withdrawal and inquiring with the patient can sometimes open up useful discussions. When a patient is unwilling or unable to provide this information, the clinician may rely more on the presenting symptoms, information obtained from family or friends, results of toxicology reports, and review of prescription drug monitoring programs. Some symptoms of opioid withdrawal include pain, anxiety/restlessness, insomnia, excessive sweating, dilated pupils, diarrhea/abdominal pain, flu-like symptoms, goose bumps, tremor, and yawning [63].

Alcohol Withdrawal Syndrome

Alcohol withdrawal is dangerous and can seriously complicate cancer treatment. In some instances, it is fatal. The first symptoms of withdrawal typically appear in the first few hours following the cessation of alcohol consumption and may consist of tremors, agitation, and insomnia. In cases of mild to moderate withdrawal, these symptoms tend to dissipate within 1–2 days without recurrence. However, in cases of severe withdrawal, autonomic hyperactivity, hallucinations, and disorientation may follow. The onset of delirium tremens marks the individual's progression from the withdrawal state to a state of delirium that represents a serious medical emergency.

Delirium tremens (DTs): characterized by agitation, hallucination, delusions, incoherence, and disorientation, typically within the first 72–96 h of withdrawal

- Occurs in approximately 5–15% of patients with alcohol withdrawal [64].
- Is self-limiting and often ends with the patient entering a deep sleep with amnesia for most of the episode.
- DTs can increase the risk of further complications in medically ill patients.

Wernicke-Korsakoff syndrome: indicative of thiamine deficiency that causes permanent cognitive impairment

- · Frequently underdiagnosed.
- Symptoms.
 - Fixed upward gaze.
 - Alcoholic neuropathy.
 - "Stocking-glove" paresthesia
 - Autonomic instability.
- Delirium encephalopathy.

Medical Treatment of Alcohol Withdrawal

While a full discussion of the pharmacological approach to alcohol withdrawal is beyond the scope of this chapter, a basic approach to its treatment is given. The use of hydration, benzodiazepines, and, in some cases, neuroleptics is appropriate for the management of alcohol withdrawal syndrome (Table 45.3). The administration of a vitamin-mineral solution is indicated to counteract the effects of malnutrition that results from the alcohol itself and poor eating habits. Thiamine 100 mg administered intramuscularly or intravenously for 3 days before switching to oral administration for the duration of treatment will prevent the development of Korsakoff syndrome and alcoholic dementia. A daily oral dose of folate 1 mg should also be given throughout the course of treatment. In cases of mild withdrawal, hydration alone may be sufficient. Benzodiazepines (lorazepam, midazolam, diazepam, and chlordiazepoxide) are the drugs of choice for the management of alcohol withdrawal because of their sedative effects (Table 45.4) [65, 66]. Careful consideration must be given to route, absorption, potency, and dose of benzodiazepine prescribed. Dose should be based upon estimated alcohol consumption and the type of detoxification setting (see below). Insufficient administration of benzodiazepines (too low dose or too rapid taper) may allow the progression of withdrawal to a state of delirium tremens. The development of seizures is life-threatening, and they may repeatedly recur in the patient while unconscious. The nonbenzodiazepine anticonvulsants are not prescribed prophylactically. In cases of severe withdrawal and confusion, neuroleptics (i.e., haloperidol 0.5-5.0 mg IV every 8 h) are added to the treatment regimen. Commonly, alcoholic patients report to the hospital either intoxicated or in the early stages of withdrawal. From a surgical perspective, serious complications can arise from the presence of alcohol

 Table 45.3
 Guidelines for the treatment of alcohol withdrawal

Continual close monitoring of withdrawal status

Utilization of benzodiazepines

Taper dose slowly (generally not by more than 25% per 24-h period) Administration of thiamine 100 mg IM or IV qid Administration of folate 1 mg po qid Monitor for signs of the potential onset of delirium tremens

Consideration should be given to a loading dose of phenytoin for patients with a history of withdrawal seizures or for patients in whom seizures are likely (i.e., patients with brain metastases)

 Table 45.4
 Types and characteristics of benzodiazepines for treatment of alcohol withdrawal

Drug	Dose	Duration of action	Half-life (h)
Chlordiazepoxide	25–100 mg every 3 h IV	Short	5-30
Diazepam	10–20 mg every 1–4 h IV	Short	20–100
Lorazepam	1–2 mg every 1–4 h IV	Intermediate	10–20
Midazolam	1–5 mg every 1–2 h IV	Very short	1–4

withdrawal, and its acute management is the primary treatment goal. Unfortunately, clinicians are frequently provided insufficient lead time to properly detoxify the patient prior to surgery (typically less than 24 h), and the patient is at an increased risk for the postoperative development of organic mental disorders, seizure, and delirium tremens. Since alcoholic cancer patients are already at high risk for delirium postoperatively due to poor nutrition, prior head trauma, and brain injury from excessive alcohol consumption, the development of seizures and delirium tremens adds to the risk of fatality. It is important to note that since it is desirable for the patient to be alert postoperatively for ambulation and use of pulmonary toilet, the amount of sedation required for detoxification is much lower than the desired level of sedation in a nonsurgical alcoholic patient.

Considering the Therapeutic Impact of Tolerance

Patients who are active substance abusers may be tolerant to drugs administered for therapy, making pain management more difficult. The magnitude of this tolerance is never known. Therefore, it is best to begin with a conservative dose of therapeutic drug and then rapidly titrate the dose, with frequent reassessments, until the patient is comfortable [40, 41, 54]. Cancer patients and those with progressive disease can be treated with gradually increasing doses, and opioids can still be titrated to effect or toxicity with no arbitrary number of milligrams constituting a limit. Tolerance to a variety of opioid effects can be reliably observed in animal models [67], and tolerance to non-analgesic effects, such as respiratory depression and cognitive impairment [68], occurs regularly in the clinical setting. However, analgesic tolerance does not appear to routinely interfere with the clinical efficacy of opioid drugs.

Psychopharmacology Approaches

Disulfiram (Antabuse) is a pharmacological agent that has been approved by the Food and Drug Administration (FDA) since 1951 for the treatment of alcoholism. Antabuse serves as a deterrent by inducing an unpleasant physical state characterized by nausea or vomiting when alcohol is consumed, thus ideally leading to alcohol cessation [69]. The practicality and effectiveness of Antabuse is questionable however, since its use has been limited by difficulties with patient adherence for continued use of the drug [70].

There have been a number of studies shedding light on subgroups of patients who have been shown to benefit the most from treatment with Antabuse. The findings have shown that patients with the following characteristics generally
experience the most long-term benefits from Antabuse: (1) older than 40 years of age, (2) longer drinking histories, (3) socially stable, (4) highly motivated, (5) prior attendance of Alcoholics Anonymous, (6) cognitively intact, and (7) able to maintain and tolerate dependent or treatment relationships [71–73]. Further research is needed to ascertain what factors and patient characteristics will increase the likelihood of successful treatment. A greater understanding has the potential to significantly enhance clinicians' ability to select those patients who will experience optimal effectiveness.

Methadone Maintenance

Methadone maintenance therapy (MMT) is superior to illegal heroin use, in part, because the extreme highs and lows felt by heroin users (related to the waxing and waning of serum heroin levels) are avoided by the long-acting properties of methadone. The term "agonist blockade" was coined to describe the phenomenon of significantly limited or blunted effects after administration of "usual" doses of muopioid agonists to subjects on high-dose methadone (e.g., 80–120 mg/day).

In humans, all opioids suppress the hypothalamicpituitary-adrenal (HPA) axis when given acutely, and this effect persists during chronic, intermittent exposure to short-acting opioids during chronic cycles of heroin addiction [74].

The endogenous mu-opioid receptor-mediated opioid system in humans appears to constitutively provide tonic inhibition of the HPA axis [74, 75]. Thus, administration of mu-opioid receptor antagonists to healthy human volunteers leads to activation of the HPA axis [75–79]. Similarly, the HPA axis is activated in opioid withdrawal (for which clonidine may be helpful [62]), or with administration of mu-opioid receptor antagonists to opioid-dependent individuals, or during acute cocaine or alcohol consumption [75, 78].

Kreek and colleagues [80] proposed that suppression of the HPA axis through administration of intermittent or binge-type short-acting opioids (e.g., heroin) and then with repeated alternating short cycling of suppression (e.g., with heroin administration), followed by activation (e.g., with heroin withdrawal [i.e., just before next dose]), may lead to and/or exacerbate atypical responsivity to stress/stressors, as well as addictive-type behavior (with resultant selfadministration/relapse). Adequate methadone maintenance treatment permits normalization of the HPA axis-including response to a chemically induced stress of metyrapone challenge [81, 82]. In an optimal situation, stabilized methadone-maintained former heroin addicts treated in highquality methadone maintenance treatment programs (e.g., associated with psychosocial interventions) with effective methadone doses experience the following: markedly

reduced drug craving; reduced or eliminated heroin use; improved or normalized stress-responsive hypothalamicpituitary-adrenal axis; and reproductive, gastrointestinal, and immunologic functions with relatively normal responses to acute pain [83, 84].

Buprenorphine and Naltrexone

Two other therapies used in the medication-assisted treatment (MAT) of those with opioid addiction and alcoholism are not without their complexities if they are to be used in people with cancer.

Buprenorphine is a partial opioid agonist that has significantly advanced MAT for opioid addiction on an international level. Available as a pill, sublingual film (with and without naltrexone), and as an implant for addiction treatment, its use in people with cancer can complicate the treatment of pain in the setting of disease progression and require dose escalation that could "bump up against" the drug's ceiling effect or in the treatment of acute pain requiring the use of a pure mu-agonist. However, there are also reports of the successful use of oral and transdermal buprenorphine for chronic and breakthrough cancer pain in nonaddicts [85]. If a person with cancer also has a history of opioid addiction and is to be managed with continuation of their buprenorphine treatment, consultation should be sought from an addiction medicine expert (who also has the appropriate certification to prescribe it where necessary). Options for managing pain exacerbations may include increasing the total daily dose and dividing the buprenorphine dose into 6- to 8-hour intervals, adding and titrating a short-acting and potent mu-agonist such as fentanyl, or discontinuing and switching buprenorphine to a full mu-opioid agonist. Emerging evidence supports that patients with opioid use disorders continue with buprenorphine treatment while the acute pain is being managed [86].

The oral opioid antagonist naltrexone is used to treat alcohol cravings and opioid addiction and is also available as a monthly depot injection for addiction treatment. While ultralow-dose naltrexone has been used to augment opioids for cancer-related pain and for the treatment of side effects such as constipation, little has been written about the use of this therapy for addiction treatment in people with cancer. While one can imagine antagonist therapy having a role in, for example, people surviving cancer who struggle with addiction (and in whom pain severe enough to require opioids is not part of the clinical picture) and in those with pain and with active disease, its role is limited. There is a paucity of data and direct clinical experience on which specific recommendations might be made. These medications can cause difficulties for the ED professionals as they might, for instance, need to intervene for acute pain in the ED setting,

the management of which is complex in persons on antagonist therapy.

Selecting Appropriate Drugs and Route of Administration for the Symptom and Setting

The use of long-acting analgesics in sufficient amounts may help to minimize the number of rescue doses needed, lessen cravings, and decrease the risk of abuse of prescribed medications, given the possible difficulty of using short-acting formulations in patients with substance abuse histories. Rather than being overly concerned regarding the choice of drug or route of administration, the prescription of opioids and other potentially abusable drugs should be carried out with limits and guidelines [21–23].

Many clinicians now respond to particularly high-opioiddose requirements with rotation to another opioid. This practice capitalizes on incomplete cross-tolerance, or the unique pharmacology of methadone in particular, to bring doses down while maintaining or improving efficacy and changing the balance of efficacy to toxicity [87, 88]. Some clinicians set arbitrary dose limits for the various opioids. Others stop using certain opioids they perceive as of higher risk or street value. Still others became so disillusioned as to stop using opioids altogether.

Recognizing Specific Drug Abuse Behaviors

In an effort to monitor the development of aberrant drugtaking behaviors, all patients who are prescribed potentially abusable drugs must be evaluated over time. This is particularly true for those patients with a remote or current history of drug abuse, including alcohol abuse. Should a high level of concern exist regarding such behaviors, frequent visits and regular assessments of significant others who can contribute information regarding the patient's drug use may be required. To promote early recognition of aberrant drugrelated behaviors, it may also be necessary to have patients with histories of recent active abuse to submit urine specimens for regular screening of illicit, or licit but nonprescribed, drugs. When informing the patient of this approach, explain that it is a method of monitoring that helps the clinician keep the patient safe and provides a foundation for aggressive symptom-oriented treatment, thus enhancing the therapeutic alliance with the patient [21-23].

Using Nondrug Approaches as Appropriate

The most effective psychotherapeutic treatment approach with medically ill people appears to be one that focuses on the development of effective coping skills, relapse prevention, and, most importantly, treatment compliance. Alcohol or the specific substance being abused represents one of the dependent patient's primary, albeit maladaptive, coping tools. As a result, the improvement of coping skills in these individuals is critical. When compounded with the stress associated with cancer, substance abuse cessation can be overwhelming and contribute to noncompliance and discontinuation of treatment. Teaching specific, illness-related coping methods with an emphasis upon containing episodes of consumption is essential. Further, the recognition and treatment of anxiety and depression may decrease the patients' need and desire for alcohol or substances. As an alternative to the abstinence approach, a harm reduction with crisis intervention as a central component should be utilized. The fundamental aims of this approach are enhancement of social support, maximization of treatment compliance, and containing harm associated with episodic relapses. Further, minimizing the frequency and intensity of the patients' use and consumption is the broad goal of treatment. Therefore, further damage to the patient will be reduced as well as the facilitation of treatment compliance.

Other psychotherapeutic approaches beneficial for this population are support groups and 12-step programs. The problem lies in that traditional 12-step groups are based on an abstinence-only policy. This poses a problem for patients who are being treated with opioids for pain-related syndromes. More recently, support groups have been tailored for this specific population.

Many nondrug approaches exist to assist patients in coping with chronic pain in advanced illness. Such educational interventions may include relaxation techniques, ways of thinking of and describing the experience of pain, and methods of communicating physical and emotional distress to staff members. Although nondrug interventions are adjuvants to management, they should not be perceived as substitutes for drugs targeting pain or other physical or psychological symptoms [21–23].

Developing the Treatment Plan

Inpatient Management Plan

In designing the inpatient management of an actively abusing patient with advanced illness, it is helpful to use structured treatment guidelines. Although the applicability of these guidelines may vary from setting to setting, they provide a set of strategies that can ensure the safety of the patient and staff, control possible manipulative behaviors, allow for supervision of illicit drug use, enhance appropriate use of medications for pain and symptom control, and communicate an understanding of pain and substance abuse management [21–23]. Under certain circumstances, such as actively abusing patients who are scheduled for surgery, patients should be admitted several days in advance, when possible, to allow for stabilization of the drug regimen. This time can also be used to avoid withdrawal and to provide an opportunity to assess whether modifications of an established plan are necessary [21, 22].

Once established, the structured treatment plan for the management of active abuse must proceed conscientiously. In an effort to assess and manage symptoms, frequent visits are usually necessary. It is also important to avoid drug with-drawal, and, to the extent possible, prescribed drugs for symptom control should be administered on a regularly scheduled basis. This helps to eliminate repetitive encounters with staff that center on the desire to obtain drugs [21–23].

Treatment management plans must be designed to represent the clinician's assessment of the severity of drug abuse. Open and honest communication between clinician and patient to stress that the guidelines were established in the best interest of the patient is often helpful. However, in cases where patients are unable to follow these guidelines despite repeated interventions from the staff, discharge should be considered. Clinicians should discuss this decision for patient discharge with the staff and administration while considering the ethical and legal ramifications of this action [21-23].

Outpatient Management Plan

Alternative guidelines may be used in the management of the actively abusing patient with advanced illness who is being treated on an outpatient basis. In some instances, the treatment plan can be coordinated with referral to a drug rehabilitation program. However, patients who are facing end-of-life issues may have difficulty participating in such programs. Using the following approaches may be helpful for managing the complex and more difficult-to-control aspects of care.

Case Study

A 36-year-old white male with stage IV lung cancer (Pancoast tumor) that was locally advanced and widely metastatic presented late after a 35-lb weight loss. His sister had died of the same disease at age 35, and he had a history of significant substance abuse and drug dealing. The patient complained of out-of-control pain and lack of willingness of any local providers to prescribe pain medication. The patient was inflexible about acceptance of any other treatments (i.e., nerve block, epidural) other than OxyContinTM. The patient's pain was 10/10 from brachial plexopathy with mixed neuropathic/somatic/visceral components.

Patient was titrated to effect over time. The maximum dose reached 800 mg bid at the time of death with 100 mg liquid MS04 q1h rescues. Although the patient was dying, structured management was required because of his history. The structured management plan was as follows: hospice nurses delivered one-day supply, unscheduled visits for pill counts, urine screens, and a reliable family member was identified to lock up pain medication supply.

The patient settled down and with renewed trust was willing to add nortriptyline, which helped with neuropathic pain, as well as steroids for nausea, cachexia, and fatigue.

Recommendations for Prescribing

Patients who are actively abusing must be seen weekly to build a good rapport with staff and afford evaluation of symptom control and addiction-related concerns. Frequent visits allow the opportunity to prescribe small quantities of drugs, which may decrease the temptation to divert and provide a motive for not missing appointments [21–23].

Procedures for prescription loss or replacement should be explicitly explained to the patient, with the stipulation that no renewals will be given if appointments are missed. The patient should also be informed that any dose changes require prior communication with the clinician. Additionally, clinicians who are covering for the primary care provider must be advised of the guidelines that have been established for each patient with a substance abuse history to avoid conflict and disruption of the treatment plan [21–23].

Twelve-Step Programs

Depending on the patient's stage of advanced illness and functional status, the clinician may consider referring the patient to a 12-step program with the stipulation that attendance be documented for ongoing prescription purposes. If the patient has one, the clinician may contact the patient's sponsor, depending on the stage of illness and individual capabilities, in an effort to disclose the patient's illness, and that medication is required in the treatment of the illness. This contact will also help decrease the risk of stigmatizing the patient as being noncompliant with the ideals of the 12-step program [21–23]. If the patient is unable to participate in a 12-step program, other psychosocial and/or spiritual team members can provide care that supports sobriety.

Urine Drug Testing and Prescription Drug Monitoring Programs

One of the most commonly utilized risk management tools in chronic non-cancer pain management and adherence monitoring/sobriety in addiction treatment is urine drug testing (UDT). Depending on the method employed, UDT can be used to gauge whether the patient is adherent to their prescribed medication, whether or not they are also taking nonprescribed licit medications, and/or whether they are using illicit drugs and alcohol [89]. Indeed, one study in which primary care doctors were taught to employ a "menu" of risk management techniques (including UDT) and then studied over time to examine their use of them found that UDT was the most commonly retained practice element on 6-month follow-up [90].

UDT is underutilized in the management of cancer-related pain with opioids [91]. Perhaps oncologic ED professionals, due to their lack of familiarity with the evolution in methods and mind-set that had occurred in the laboratory and clinic in the last decade, think of UDT in only its forensic incarnation. In that view, UDT is a means out of finding if "bad people" are "doing bad things," as seen in a prior chart review study [92]. Thus, because of this stigma, they are fearful that introducing UDT to their patients and integrating it into patient management will be offensive, and they may lack an approach for discussing it in a non-shaming manner with their patients. However, recent research in addiction treatment suggests that UDT can be utilized with patients in a non-stigmatizing manner when clinicians take the perspective that patients with substance abuse problems deserve help, and UDT is seen as a supportive tool for the patient's benefit [93]. The forensic method, from which more modern clinical testing sprung, tends to rely on immunoassay (IA) testing which offers fast but only class-level (not drugspecific) results with high cutoffs. It is meant to detect recent use of classes of drugs that would impair, for example, a truck driver from driving. Cutoffs are high because of the legal and other consequences that could follow and to avoid falsely accusing people. In recent years, gas chromatography and liquid chromatography with mass spectrometry are capable of giving highly accurate drug-specific results and return them in a timely fashion (1-2 days as opposed to 10 days-2 weeks). Such results can be used to determine whether a patient is misusing a range of drugs or alcohol and gauge their adherence with specific medication and controlled substance regimens. More data is needed in this area [91], but the use of UDT for those with pain and/or substance use disorder is well documented [94–98].

Oncologic personnel in the ED should routinely access prescription drug monitoring program (PDMP) prior to prescribing a scheduled medication. Checking PDMPs can identify patients who may be receiving overlapping prescriptions, obtaining prescriptions from multiple prescribers, or misusing their scheduled medications [99]. A point that is often overlooked is that patients in oncologic settings can benefit from the clinician's use of PDMPs as much as in any chronic pain setting. Patients being treated for cancer are more likely than with some other chronic pain conditions to have multiple prescribers. Coordinating care can be a challenge, and patients are at risk for being overmedicated. Although not intended for patients with cancer, clinicians should be aware of the Center for Disease Control and Prevention guidelines [100], which recommend checking the PDMP prior to initiating a scheduled medication and at least every 3 months thereafter.

Family Sessions and Meetings

The clinician, in an effort to increase support and function, should involve family members and friends in the treatment plan. These meetings allow the clinician and other team members to become familiar with the family and additionally help the team identify family members who are using illicit drugs. Offering referral of these identified family members to drug treatment can be portrayed as a method of gathering support for the patient. The patient should also be prepared to cope with family members or friends who may attempt to buy or sell the patient's medications. These meetings will also assist the team in identifying dependable individuals who can serve as a source of strength and support for the patient during treatment [21-23]. These published guidelines generally advocate an approach to UDT based on risk stratification (i.e., frequency of testing and choice of methods are aimed at matching the approach to the level of risk of abuse, addiction, and diversion in an individualized way to each patient). Such an approach seeks to maximize the benefit of testing while also managing cost. As oncologic professionals learn to integrate UDT into treatment of the person with cancer and addiction or the management of chronic opioid therapy, there is no reason to think that the adaptation of a similar approach might not be a reasonable way to proceed.

Conclusion

Treating oncologic patients with chronic pain and substance use disorders is both complicated and challenging, as each can significantly complicate the other. Whether our patients respond to cancer treatments or have life-limiting disease, we can no longer justify high-dose opioid therapy in a vacuum without trying to assess and manage addiction and abuse behaviors. Using a treatment plan that involves a team approach to recognize and respond to these complex needs is the optimum treatment strategy. While pain management may remain challenging even when all treatment plan procedures are implemented, the healthcare team's goal should be providing the highest level of pain management for all patients with substance use disorders.

References

- Cicero TJ, Surratt HL, Kurtz S, Ellis MS, Inciardi JA. Patterns of prescription opioid abuse and comorbidity in an aging treatment population. J Subst Abus Treat. 2012;42(1):87–94.
- Minozzi S, Amato L, Davoli M. Development of dependence following treatment with opioid analgesics for pain relief: a systematic review. Addiction. 2013;108(4):688–98.
- Passik SD. Responding rationally to recent report of abuse/diversion of Oxycontin. J Pain Symptom Manag. 2001;21(5):359.
- Cicero TJ, Ellis MS, Kasper ZA. Polysubstance use: a broader understanding of substance use during the opioid crisis. Am J Public Health. 2020;110(2):244–50.
- Lowery AE, Starr T, Dhingra LK, Rogak L, Hamrick-Price JR, Farberov M, et al. Frequency, characteristics and correlates of pain in a pilot study of colorectal cancer survivors 1–10 years posttreatment. Pain Med. 2013;14(11):1673–80.
- Modesto-Lowe V, Girard L, Chaplin M. Cancer pain in the opioid-addicted patient: can we treat it right? J Opioid Manag. 2012;8(3):167–75.
- Jairam V, Yang DX, Yu JB, Park HS. Emergency department visits for opioid overdoses among patients with cancer. J Natl Cancer Inst. 2020;112(9):938–43.
- Reyes-Gibby CC, Anderson KO, Todd KH. Risk for opioid misuse among emergency department cancer patients. Acad Emerg Med. 2016;23(2):151–8.
- Madden K, Reddy AS, De La Cruz MJ, Liu DD, Bruera E. Patterns of storage, use, and disposal of prescription opioids by parents of children with cancer. J Pain Symptom Manag. 2020;59(2):320–6.
- Cantrill SV, Brown MD, Carlisle RJ, Delaney KA, Hays DP, Nelson LS, et al. American College of Emergency Physicians Opioid Guideline Writing Panel. Clinical policy: critical issues in the prescribing of opioids for adult patients in the emergency department. Ann Emerg Med. 2012;60(4):499–525.
- De Ruddere L, Craig KD. Understanding stigma and chronic pain: a-state-of-the-art review. Pain. 2016;157(8):1607–10.
- Kelley JM, Kraft-Todd G, Schapira L, Kossowsky J, Riess H. The influence of the patient-clinician relationship on healthcare outcomes: a systematic review and meta-analysis of randomized controlled trials. PLoS One. 2014;9(4):e94207.
- Patel PM, Goodman LF, Knepel SA, Miller CC, Azimi A, Phillips G, et al. Evaluation of emergency department management of opioid-tolerant cancer patients with acute pain. J Pain Symptom Manag. 2017;54(4):501–7.
- Kirsh KL, Peppin JF, Coleman J. Characterization of prescription opioid abuse in the United States: focus on route of administration. J Pain Palliat Care Pharmacother. 2012;26(4):348–61.
- Kircher S, Zacny J, Apfelbaum SM, Passik SD, Kirsh KL, Burbage M, et al. Understanding and treating opioid addiction in a patient with cancer pain. J Pain. 2011;12(10):1025–31.
- Substance Abuse and Mental Health Services Administration. Key substance use and mental health indicators in the United States: Results from the 2018 National Survey on Drug Use and Health (HHS Publication No. PEP19–5068, NSDUH Series H-54).

Rockville: Center for Behavioral Health Statistics and Quality, Substance Abuse and Mental Health Services Administration; 2019. Retrieved from https://www.samhsa.gov/data/SAMHSA.

- Gfroerer J, Brodsky M. The incidence of illicit drug use in the United States 1962–1989. Br J Addict. 1992;87(9):1345–51.
- Muirhead G. Cultural issues in substance abuse treatment. Patient Care. 2000;5:151–9.
- Regier DA, Myers JK, Kramer M, Robins LN, Blazer DG, Hough RL, et al. The NIMH Epidemiologic Catchment Area program. Historical context, major objectives, and study population characteristics. Arch Gen Psychiatry. 1984;41(10):934–41.
- 20. Substance Abuse and Mental Health Services Administration. National Survey on drug use and health: illicit drug use among older adults. Substance Abuse and Mental Health Services Administration: Rockville; 2011.
- Kirsh KL, Passik SD. Patients with a history of substance abuse. In: Smith HS, editor. Opioid therapy in the 21st century. 2nd ed. New York: Oxford University Press; 2013. p. 255–62.
- 22. Passik SD, Kirsh KL. What approaches should be used to minimize opioid diversion and abuse in palliative care? In: Goldstein N, Morrison S, editors. Evidence-based practice of palliative medicine. Philadelphia: Elsevier; 2013. p. 87–92.
- Rzetelny A, Ruehle M, Miller N, Kirsh KL, Passik SD. Substanceuse disorders in older adults with chronic cancer pain. In: Holland JC, Wiesel TW, Nelson CJ, Roth AJ, Alici Y, editors. Geriatric psycho-oncology: a quick reference on the psychosocial dimensions of cancer symptom management. New York: Oxford; 2015. p. 79–86.
- 24. Bruera E, Moyano J, Seifert L, Fainsinger RL, Hanson J, Suarez-Almazor M. The frequency of alcoholism among patients with pain due to terminal cancer. J Pain Symptom Manag. 1995;10(8):599–603.
- 25. Winick C. Maturing out of narcotic addiction. Bull Narc. 1962;14(1):1–7.
- Patterson TL, Jeste DV. The potential impact of the baby-boom generation on substance abuse among elderly persons. Psychiatr Serv. 1999;50(9):1184–8.
- Colliver JD, Compton WM, Gfroerer J, Condon T. Projecting drug use among aging baby boomers in 2020. Ann Epidemiol. 2006;16(4):257–65.
- Manchikanti L, Singh A. Therapeutic opioids: a ten-year perspective on the complexities and complications of the escalating use, abuse, and nonmedical use of opioids. Pain Physician. 2008;11(2 Suppl):S63–8.
- Taylor MH, Grossberg GT. The growing problem of illicit substance abuse in the elderly: a review. Prim Care Companion CNS Disord. 2012;14(4):PCC.11r01320.
- Borgelt LM, Franson KL, Nussbaum AM, Wang GS. The pharmacologic and clinical effects of medical cannabis. Pharmacotherapy. 2013;33(2):195–209.
- 31. Substance Abuse and Mental Health Services Administration, Center for Behavioral Health Statistics and Quality. Drug abuse warning network (DAWN). Emergency department visits involving illicit drug use by older adults: 2008. Rockville: SAMHSA. p. 2010.
- 32. Substance Abuse and Mental Health Services Administration, Center for Behavioral Health Statistics and Quality. Drug abuse warning network (DAWN). The DAWN report: drug-related emergency department visits involving pharmaceutical misuse and abuse by older adults. Rockville: SAMHSA; 2010.
- Rockett IR, Putnam SL, Jia H, Smith GS. Declared and undeclared substance use among emergency department patients: a population-based study. Addiction. 2006;101(5):706–12.
- 34. Substance Abuse and Mental Health Services Administration. Treatment episode data set (TEDS): older adult admissions reporting alcohol as a substance of abuse: 1992 and 2009. Rockville:

Substance Abuse and Mental Health Services Administration; 2011.

- 35. Substance Abuse and Mental Health Services Administration. Treatment Episode Data Set (TEDS). Changing substance abuse patterns among older admissions: 1992 and 2008. Rockville: Substance Abuse and Mental Health Services Administration (SAMHSA); 2010.
- Sjøgren P, Ekholm O, Peuckmann V, Grønbaek M. Epidemiology of chronic pain in Denmark: an update. Eur J Pain. 2009;13(3):287–92.
- Manchikanti L, Damron KS, McManus CD, Barnhill RC. Patterns of illicit drug use and opioid abuse in patients with chronic pain at initial evaluation: a prospective, observational study. Pain Physician. 2004;7(4):431–7.
- Morasco B, Dobscha S. Prescription medication misuse in substance use disorder in VA primary care patients with chronic pain. Gen Hosp Psychiatry. 2008;30(2):93–9.
- Gfroerer J, Penne M, Pemberton M, Folsom R. Substance abuse treatment need among older adults in 2020: the impact of the aging baby-boom cohort. Drug Alcohol Depend. 2003;69(2):127–35.
- Hamrick JR, Passik SD, Kirsh KL. Substance abuse issues in palliative care. In: Berger AM, Shuster JL, Von Roenn JH, editors. Principles and practice of palliative care and supportive oncology. 4th ed. Philadelphia: Lippincott Williams & Wilkins; 2013. p. 575–89.
- Passik SD, Portenoy RK. Substance abuse issues in palliative care. In: Berger A, Portenoy R, Weissman D, editors. Principles and practice of supportive oncology. Philadelphia: Lippincott-Raven; 1998. p. 513–24.
- Glajchen M, Fitzmartin RD, Blum D, Swanton R. Psychosocial barriers to cancer pain relief. Cancer Pract. 1995;3(2):76–82.
- Ramer L, Richardson JL, Cohen MZ, Bedney C, Danley KL, Judge EA. Multimeasure pain assessment in an ethnically diverse group of patients with cancer. J Transcult Nurs. 1999;10(2):94–101.
- 44. Ward SE, Goldberg N, Miller-McCauley V, Mueller C, Nolan A, Pawlik-Plank D, et al. Patient-related barriers to management of cancer pain. Pain. 1993;52(3):319–24.
- Passik SD, Webster L, Kirsh KL. Pseudoaddiction revisited: a commentary on clinical and historical considerations. Pain Manage. 2011;1(3):239–48.
- Argoff CE. Clinical implications of opioid pharmacogenetics. Clin J Pain. 2010;26(1 Suppl):S16–20.
- Passik SD, Kirsh KL, Donaghy KB, Portenoy RK. Pain and aberrant drug-related behaviors in medically ill patients with and without histories of substance abuse. Clin J Pain. 2006;22(2):173–81.
- Fleming MF, Balousek SL, Klessig CL, Mundt MP, Brown DD. Substance use disorders in a primary care sample receiving daily opioid therapy. J Pain. 2007;8(7):573–82.
- Passik SD, Portenoy RK, Ricketts PL. Substance abuse issues in cancer patients part 1: prevalence and diagnosis. Oncology. 1998;12(4):517–21.
- Steindler EM, Wilford BB. Clarification and standardization of substance abuse terminology. JAMA. 1988;259(4):555–7.
- Khantzian EJ, Treece C. DSM-III psychiatric diagnosis of narcotic addicts. Arch Gen Psychiatry. 1985;42(11):1067–71.
- Dunbar SA, Katz NP. Chronic opioid therapy for nonmalignant pain in patients with a history of substance abuse: report of 20 cases. J Pain Symptom Manag. 1996;11(3):163–71.
- Gonzales GR, Coyle N. Treatment of cancer pain in a former opioid abuser: fears of the patient and staff and their influences on care. J Pain Symptom Manag. 1992;7(4):246–9.
- Macaluso C, Weinberg D, Foley KM. Opioid abuse and misuse in a cancer pain population (abstract). J Pain Symptom Manag. 1988;3:S24–31.

- Kemp C. Managing chronic pain in patients with advanced disease and substance related disorders. Home Healthc Nurse. 1996;14(4):255–61.
- Savage SR. Addiction in the treatment of pain: significance, recognition, and management. J Pain Symptom Manag. 1993;8(5):265–77.
- Passik SD, Portenoy RK, Ricketts PL. Substance abuse issues in cancer patients part 2: evaluation and treatment. Oncology. 1998;12(5):729–34.
- Passik SD, Kirsh KL, Casper D. Addiction-related assessment tools and pain management: instruments for screening, treatment planning, and monitoring compliance. Pain Med. 2008;9(S2):S145–66.
- Smith HS, Kirsh KL. Identifying and managing the risk of opioid misuse. Therapy. 2009;6(5):685–93.
- Helzer JE, Pryzbeck TR. The co-occurrence of alcoholism with other psychiatric disorders in the general population and its impact on treatment. J Stud Alcohol. 1988;49(3):219–24.
- Regier DA, Farmer ME, Rae DS, Locke BZ, Keith SJ, Judd LL, Goodwin FK. Comorbidity of mental disorders with alcohol and other drug abuse. Results from the Epidemiologic Catchment Area (ECA) Study. JAMA. 1990;264(19):2511–8.
- Patel B, Kosten TR. Keeping up with clinical advances: opioid use disorder. CNS Spectr. 2019;24(S1):14–24.
- Kosten TR, Baxter LE. Effective management of opioid withdrawal symptoms: a gateway to opioid dependence treatment. Am J Addict. 2019;28(2):55–62.
- 64. Maxmen JS, Ward NG. Substance-related disorders. In: Essential psychopathology and its treatment. 2nd ed. New York: WW. Norton and Company; 1995. p. 132–72.
- Erstad BL, Cotugno CL. Management of alcohol withdrawal. Am J Health Syst Pharm. 1995;52(7):697–709.
- Newman JP, Terris DJ, Moore M. Trends in the management of alcohol withdrawal syndrome. Laryngoscope. 1995;105(1):1–7.
- 67. Ling GSF, Paul D, Simantov R, Pasternak GW. Differential development of acute tolerance to analgesia, respiratory depression, gastrointestinal transit and hormone release in a morphine infusion model. Life Sci. 1989;45(18):1627–36.
- Bruera E, Macmillan K, Hanson JA, MacDonald RN. The cognitive effects of the administration of narcotic analgesics in patients with cancer pain. Pain. 1989;39(1):13–6.
- Suh JJ, Pettinati HM, Kampman KM, O'Brien CP. The status of disulfiram: a half of a century later. J Clin Psychopharmacol. 2006;26(3):290–302.
- Weinrieb RM, O'Brien CP. Current research in the treatment of alcoholism in liver transplant recipients. Liver Transpl Surg. 1997;3(3):328–36.
- Banys P. The clinical use of disulfiram (Antabuse): a review. J Psychoactive Drugs. 1988;20(3):243–61.
- 72. Fuller RK, Gordis E. Does disulfiram have a role in alcoholism treatment today? Addiction. 2004;99(1):21–4.
- Hughes JC, Cook CC. The efficacy of disulfiram: a review of outcome studies. Addiction. 1997;92(4):381–95.
- Kreek MJ, Oratz M, Rothschild MA. Hepatic extraction of longand short- acting narcotics in the isolated perfused rabbit liver. Gastroenterology. 1978;75(1):88–94.
- Schluger JH, Ho A, Borg L, Porter M, Maniar S, Gunduz M, et al. Nalmefene causes greater hypothalamic-pituitary-adrenal axis activation than naloxone in normal volunteers: implications for the treatment of alcoholism. Alcohol Clin Exp Res. 1998;22(7):1430–6.
- Culpepper-Morgan JA, Inturrisi CE, Portenoy RK, Foley K, Houde RW, Marsh F, et al. Treatment of opioid-induced constipation with oral naloxone: a pilot study. Clin Pharmacol Ther. 1992;52(1):90–5.

- Culpepper-Morgan JA, Kreek MJ. Hypothalamic-pituitary-adrenal axis hypersensitivity to naloxone in opioid dependence: a case of naloxone-induced withdrawal. Metabolism. 1997;46(2):130–4.
- King AC. Role of naltrexone in initial smoking cessation: preliminary findings. Alcohol Clin Exp Res. 2002;26(12):1942–4.
- Rosen MI, McMahon TJ, Woods SW, Pearsall HR, Kosten TR. A pilot study of dextromethorphan in naloxone-precipitated opiate withdrawal. Eur J Pharmacol. 1996;307(3):251–7.
- Schluger JH, Borg L, Ho A, Kreek MJ. Altered HPA axis responsivity to metyrapone testing in methadone maintained former heroin addicts with ongoing cocaine addiction. Neuropsychopharmacology. 2001;24(5):568–75.
- Kreek MJ. Medical safety and side effects of methadone in tolerant individuals. JAMA. 1973;223(6):665–8.
- Kreek MJ. Plasma and urine levels of methadone. Comparison following four medication forms used in chronic maintenance treatment. N Y State J Med. 1973;73(23):2773–7.
- 83. Kling MA, Carson RE, Borg L, Zametkin A, Matochik JA, Schluger J, et al. Opioid receptor imaging with positron emission tomography and [(18)F]cyclofoxy in long-term, methadone-treated former heroin addicts. J Pharmacol Exp Ther. 2000;295(3):1070–6.
- Kreek MJ. Methadone-related opioid agonist pharmacotherapy for heroin addiction: history, recent molecular and neurochemical research and future in mainstream medicine. Ann N Y Acad Sci. 2000;909:186–216.
- Atkinson TJ, Fudin J, Pandula A, Mirza M. Medication pain management in the elderly: unique and underutilized analgesic treatment options. Clin Ther. 2013;35(11):1669–89.
- Alford DP, Compton P, Samet JH. Acute pain management for patients receiving maintenance methadone or buprenorphine therapy. Ann Intern Med. 2006;144(2):127–34.
- Wirz S, Wartenberg HC, Elsen C, Wittmann M, Diederichs M, Nadstawek J. Managing cancer pain and symptoms of outpatients by rotation to sustained-release hydromorphone: a prospective clinical trial. Clin J Pain. 2006;22(9):770–5.
- Zimmermann C, Seccareccia D, Booth CM, Cottrell W. Rotation to methadone after opioid dose escalation: how should individualization of dosing occur? J Pain Palliat Care Pharmacother. 2005;19(2):25–31.
- Passik SD, Kirsh KL. Ethical considerations in urine drug testing. J Pain Palliat Care Pharmacother. 2011;25(3):265–6.

- Brown J, Setnik B, Lee K, Wase L, Roland CL, Cleveland JM, et al. Assessment, stratification, and monitoring of the risk for prescription opioid misuse and abuse in the primary care setting. J Opioid Manag. 2011;7(6):467–83.
- Arthur JA. Urine drug testing in cancer pain management. Oncologist. 2020;25(2):99–104.
- 92. Passik SD, Schreiber J, Kirsh KL, Portenoy RK. A chart review of the ordering and documentation of urine toxicology screens in a cancer center: do they influence patient management? J Pain Symptom Manag. 2000;19(1):40–4.
- Rzetelny A, Zeller B, Miller N, Kirsh K, Passik S. Definitive LC-MS/MS drug monitoring impacts substance-use treatment planning and patient outcomes: a brief report. J Addict Med. 2016;10(6):443–7.
- 94. Chou R, Fanciullo GJ, Fine PG, Miaskowski C, Passik SD. Opioids for chronic noncancer pain: prediction and identification of aberrant drug-related behaviors: a review of the evidence for an American Pain Society and American Academy Clinical Practice Guideline. J Pain. 2009;10(2):131–46.
- Christo P, Manchikanti L, Ruan X, Bottros M, Hansen H, Solanki D, et al. Urine drug testing in chronic pain: comprehensive review. Pain Physician. 2011;14(2):123–43.
- 96. Federation of State Medical Boards of the United States, Inc. Model policy for the use of controlled substances for the treatment of pain. Dallas: Federation of State Medical Boards of the United States, Inc.; 2004.
- 97. Pesce A, Gonzales E, Almazan P, Mikel C, Latyshev S, West C, Strickland J. Medication and illicit substance use analyzed using liquid chromatography tandem mass spectrometry (LC-MS/MS) in a pain population. J Anal Bioanal Techniques. 2012;3:3.
- Trescot AM, Boswell MV, Atluri SL, Hansen HC, Deer TR. Opioid guidelines in the management of chronic non-cancer pain. Pain Physician. 2006;9(1):1–39.
- 99. Wilson MP, Cucciare MA, Porter A, Chalmers CE, Mullinax S, Mancino M, Oliveto AH. The utility of a statewide prescription drug-monitoring database vs the current opioid misuse measure for identifying drug-aberrant behaviors in emergency department patients already on opioids. Am J Emerg Med. 2020;38(3):503–7.
- Dowell D, Haegerich TM, Chou R. CDC guideline for prescribing opioids for chronic pain—United States, 2016. JAMA. 2016;315(15):1624–45.

Radiology

Jihoon Lim, Keith D. Herr, and Tarek N. Hanna



46

Introduction

Oncologic emergencies encompass a wide spectrum of pathology and can affect any organ system. These can be divided into metabolic, hematologic, and structural emergencies [1]. Metabolic and hematologic conditions are frequently diagnosed based on clinical and laboratory findings, often with only incidental imaging support. Structural emergencies are those arising from mass effect, tissue infiltration, tumor hemorrhage, vascular invasion with resulting occlusion or hemorrhage, and organ drainage pathway obstruction. Diagnostic imaging can also provide a "road map" for subsequent image-guided interventional and noninvasive therapies.

Imaging Modalities

There is a wide array of diagnostic imaging options available for evaluating oncologic emergencies. Selecting the appropriate imaging modality requires consideration of availability, speed, patient-specific factors, and anticipated diagnostic yield.

Plain Radiography Rapid, universally available, low-cost screening modality. Core utility is for osseous and pulmonary evaluation and to screen for intestinal obstruction and pneumoperitoneum. Particularly useful in trauma to the appendicular skeleton. Less sensitive in the diagnosis of axial and spinal trauma, although still used as a screening tool.

Computed Tomography (CT) Rapid and highly available. Mainstay in the emergent evaluation of the head, neck, chest, abdomen, and pelvis. Modern multidetector CT technology

J. Lim \cdot K. D. Herr \cdot T. N. Hanna (\boxtimes)

allows for thin X-ray beam collimation to 0.625 mm, allowing for isotropic image acquisition and post-processing capabilities, including orthogonal and curved planar reformations, as well as 3D surface rendering [2].

The diagnostic value of CT derives from its ability to discriminate tissues based on physical density, measured in Hounsfield units (HU) and displayed in gray scale on a picturing and archiving communication system (PACS) workstation. By convention, water has a density of 0 HU and appears intermediate in gray scale. Air has a density of approximately -1000 HU and appears relatively dark (hypodense or hypoattenuating). Bone and calcium have densities in the range of +1000 HU and appear relatively bright (hyperdense or hyperattenuating). Intravascular or unclotted extravascular blood, for example, has a density of around 30-45 HU but increases in attenuation to 45-70 HU as clot develops [3]. Administration of iodine-based nonionic intravenous (IV) contrast increases soft tissue and vascular conspicuity, further improving diagnostic yield. Most malignancies display predictable enhancement characteristics, and the use of IV contrast often permits accurate diagnosis. In oncologic emergencies, routine use of IV contrast is suggested for the neck, chest, abdomen, and pelvis, primarily for soft tissue evaluation. However, ED personnel evaluating female patients with pelvic pain should refrain from ordering an abdominal or pelvic CT scan without first consulting the gynecologic oncologic team, as such scans are often of low yield [4]. For the evaluation of osseous pathology, IV contrast is not usually necessary. Most intracranial emergencies are imaged with head CT without IV contrast, as opacified intracranial vessels may obscure extra-axial hemorrhage. Optimizing arterial or venous enhancement by adjusting the timing of image acquisition allows for an assessment of vascular abnormalities, such as dural venous sinus or cortical vein thrombosis (on CT venogram) or vasospasm (on CT angiogram). Recent advances in CT technology have led to increased utilization of dual-energy CT in clinical practice. Dual-energy CT assesses attenuation at two

Department of Radiology and Imaging Sciences and Department of Emergency Medicine, Emory University, Atlanta, GA, USA e-mail: tarek.hanna@emory.edu

[©] Springer Nature Switzerland AG 2021

K. H. Todd et al. (eds.), Oncologic Emergency Medicine, https://doi.org/10.1007/978-3-030-67123-5_46

different energy levels, usually at low energy (80–100 kVp) and high energy (140–150 kVp). Based on atomic interactions between imaged tissue and X-rays at low and high energy levels, it is often possible to determine the type of material in each voxel, improving tissue contrast beyond conventional single-energy CT [5]. This approach may help to distinguish between traumatic and pathologic bone fractures [6].

Magnetic Resonance Imaging (MRI) May not be accessible after-hours at many centers. MRI requires a greater degree of patient stability and compliance than does CT and entails longer imaging times. Due to the powerful magnet used in MRI, ferromagnetic metal-containing objects inside the body can pose significant risk to the patient, as they may heat up or migrate during the exam. As a result, the presence of ferromagnetic metal-containing devices within the patient is an absolute contraindication for MRI. In addition, implanted cardiac devices may malfunction in a strong magnet and are often contraindicated for MRI. The core utility for MRI in the emergent setting is for intracranial (e.g., acute stroke, tumor burden, dural venous sinus thrombosis) and spine (e.g., cord compression, cord edema, epidural tumor, and osseous involvement) evaluation. Gadolinium-based IV contrast aids in quantifying the extent of malignant disease and can potentially characterize soft tissue tumors. As with CT, IV contrast is typically not essential for osseous evaluation but may be helpful in characterizing bone invasion of a soft tissue tumor. Commonly performed MR imaging sequences for the brain include T1-weighted (T1), T2-weighted (T2), fluid-attenuated inversion recovery (FLAIR), diffusionweighted imaging (DWI), and gradient echo (GRE). For spinal imaging, typical sequences are acquired in the sagittal plane and include T1, T2, and either T2 fat-saturated or short-tau inversion recovery (STIR) images, as well as GRE and postcontrast sequences. Axial image sequences for the spine are institution-dependent but typically include T1 and T2 images. If IV contrast is administered, postcontrast T1 images are obtained using fat-saturation techniques. Signal from fat is bright (hyperintense) on MRI. When background fat signal is suppressed, it appears dark (hypointense), increasing the conspicuity of contrastenhanced malignant tissue. MRI imaging of the long bones often employs proton density (PD) sequences as an alternative to fat-saturated T2 images. Signal from blood is hypointense on GRE and demonstrates variable signal on T1 and T2 images depending on time course. As with CT angiography (CTA) or venography (CTV), MR angiography (MRA) and venography (MRV) can be used to assess for vascular pathology, although IV contrast is not always necessary.

Ultrasonography (US) Rapid, universally available, mobile, and accessible for clinician bedside usage. This soundwave-based modality is limited by image degradation that occurs at tissue interfaces with bone or air. Doppler US is ideally suited for the assessment of blood vessels and soft tissue vascularity and highly useful in procedural guidance owing to its portability and absence of radiation. US is a good tool for extremity soft tissue disease, ascites, pleural effusions, vessel patency, and biliary pathology. Some drawbacks of ultrasound include susceptibility to artifacts and dependence on operator technique and patient cooperation. (A separate chapter on emergency ultrasound follows.)

Nuclear Medicine In the emergent setting, ventilationperfusion (V/Q) imaging may be used to evaluate for pulmonary embolism. V/Q studies are reserved for patients with contraindications to CT pulmonary angiogram (CTPA), such as a history of significant adverse reaction to iodinated IV contrast agents or renal insufficiency. A tagged red blood cell scan (erythrocytes labeled with technetium-99^m) may be used to localize the general area of gastrointestinal bleeding to guide subsequent treatment. Positron emission tomography (PET) has minimal value in the emergent oncologic patient due to increased patient preparation requirements, lengthy acquisition time, and limited availability. Other nuclear medicine exams have established utility in imaging oncologic patients (e.g., whole-body bone scintigraphy); however, these are infrequently used in the emergent setting.

Fluoroscopy Useful for esophagrams and to provide procedural guidance for lumbar puncture, myelogram injection, joint aspiration, and tube/drain placement.

Interventional Radiology Interventional procedures can provide further diagnostic information, as well as guide therapies in the acute setting, most commonly with the use of conventional catheter angiography, embolization for acute hemorrhage, and tube or drain placement for relieving obstruction or abscess drainage.

Radiologic Evaluation of Oncologic Emergencies

The following is a concise review of the imaging evaluation of oncologic emergencies, highlighting important imaging characteristics of key malignancy-related conditions across a range of organ systems to serve primarily as a resource for clinicians involved in cancer care and for radiology trainees, who will invariably encounter oncologic emergencies during training and beyond.

Neurologic Emergencies

Intracranial Mass Effect, Edema, and Hemorrhage Fundamentally, malignant tumors are space-occupying masses that compete with normal anatomical structures for limited space in the brain and spinal canal. Intracranial mass effect derives from a combination of tumor volume and peritumoral edema, which together are responsible for causing various cerebral herniation patterns and hydrocephalus. Effective tumor volume can change rapidly as a consequence of necrosis, hemorrhage, and cyst formation, which contribute to mass effect (Fig. 46.1).

On unenhanced head CT, the appearance of malignant tissue is variable, depending on its histological composition. Generally, malignant tumors are slightly hyperdense relative

to white matter, and hemorrhagic or melanin-containing components are considerably denser (e.g., malignant melanoma). Cystic and necrotic components generally measure near-water attenuation depending on the presence of cellular debris or blood products. Perilesional edema results in relative hypodensity of the surrounding brain parenchyma. Especially on unenhanced CT, the presence of edema and associated mass effect may be the only indication of underlying malignancy, and further evaluation with a contrastenhanced study, ideally MRI, is indicated. Edema from malignancy is a reactive process, and the amount of edema is proportional to tumor size and rate of growth. Accordingly, a small but rapidly growing mass may present as precipitously as a larger but more indolent mass. Tumor cell lysis and treatment-related neurotoxicity represent important additional sources of edema and mass effect, which can be antici-

Fig. 46.1 Intracranial mass effect. (a) Axial postcontrast T1 brain MR image demonstrates a rim-enhancing intraparenchymal cavitary metastatic mass (arrow). (b) Coronal contrast-enhanced T1 MR image in the same patient demonstrates left-to-right midline shift (dotted line), compatible with subfalcine herniation. Uncal herniation is also shown by an arrow on image **b**. (**c**) T2 axial brain MR image depicts marked T2-hyperintense edema surrounding a cystic intraparenchymal metastatic lesion in the left parietal lobe (arrow). Note extent of edema beyond the actual mass (arrowhead). (d) Unenhanced head CT image demonstrates a left frontal lobe mass with marked surrounding hypodense edema (arrow) and local sulcal effacement and left-to-right midline shift (dotted line). There is small-volume intra-tumoral hemorrhage (arrowhead), which may have caused this patient's acute presentation



pated and prospectively managed with systemic corticosteroid administration.

Intracranial herniation resulting from mass effect can be classified as subfalcine, transtentorial, transalar, and tonsillar [7]. Transtentorial herniation can be further characterized as ascending (originating from the posterior fossa), descending (cerebral hemispheres), or uncal (temporal lobes), depending on the location of the tumor [1]. Ascending transalar herniation results from middle cranial fossa mass effect and causes superior and anterior temporal lobe displacement across the sphenoid ridge [7]. Descending transalar herniation results from frontal lobe mass effect and causes posterior and inferior displacement across the sphenoid wing [7]. Masses located in tightly confined structures, such as the posterior fossa, can result in the rapid development of clinically significant herniation.

Patients with hypervascular metastases, such as renal cell carcinoma, melanoma, thyroid cancer, and choriocarcinoma, are at highest risk for both intra-axial (within the brain substance) and extra-axial (within the epidural, subdural, or subarachnoid space) hemorrhage [8, 9]. Intracranial hemorrhage can also result from acute disseminated intravascular coagulation (DIC), to which patients with hematologic tumors are particularly predisposed [10]. Regardless of cause, intraaxial hemorrhage appears as a hyperdense mass with a variable degree of circumferential edema on unenhanced head CT. Hyperdensity in the subarachnoid, subdural, or epidural spaces indicates extra-axial hemorrhage. Contrast-enhanced brain MRI is essential for further characterizing the underlying malignancy.

Hydrocephalus Hydrocephalus is classified as noncommunicating (obstructive) and communicating (nonobstructive). Noncommunicating hydrocephalus results from obstruction of CSF flow, while communicating hydrocephalus is the result of excess CSF production or decreased resorption at the arachnoid villi. The foramen of Monro, aqueduct of Sylvius, and the fourth ventricle are anatomically prone to obstruction by the presence of an adjacent primary or metastatic mass [9]. Specifically, pineal metastases or primary pineal neoplasms have a particular association with hydrocephalus given the location of the gland within the narrow channel between the third ventricle and the superior cerebellar cistern [11]. The primary feature of communicating (nonobstructive) hydrocephalus on unenhanced CT is global ventricular enlargement. In noncommunicating (obstructive) hydrocephalus, there is disproportionate enlargement of the lateral ventricles; the third ventricle may also be disproportionately enlarged depending on the anatomic level of obstruction. In acute hydrocephalus, increased ventricular pressure can result in transependymal CSF accumulation, resulting in a low-density appearance to the immediate periventricular white matter.

Leptomeningeal Metastatic Disease The pia and arachnoid mater are interconnected, thin, weblike tissue that envelops the brain parenchyma and together comprise the leptomeninges; leptomeningeal metastatic disease (also referred to as leptomeningeal carcinomatosis) is characterized by the deposition of tumor along these membranes. These tumoral cells can subsequently impede CSF resorption by obstructing the arachnoid villi, leading to communicating hydrocephalus [9]. Leptomeningeal metastatic disease portends a dim prognosis, with median survival in the range of 2-3 months [12]. It is important to note that up to 40% of patients with leptomeningeal metastatic disease may have normal unenhanced CT, and in an additional 25% of cases, leptomeningeal metastatic disease is indistinguishable from intraparenchymal disease [13]. Unenhanced MRI findings include high FLAIR signal within the cerebral sulci, cerebellar folia, and cisterns and communicating hydrocephalus [14]. Contrast-enhanced MRI findings include linear or nodular enhancement within the sulci, cisterns, and ventricles and along the cranial nerves [1]. The most common primary solid malignancies associated with leptomeningeal metastatic disease are breast and lung [13].

Dural Venous Sinus Thrombosis There are many benign causes of dural venous sinus thrombosis (DVST), including oral contraceptive use, pregnancy, thrombophilic disorders (such as factor V Leiden, protein C/S deficiency), recent immobilization, and infection, such as meningitis or mastoiditis [15]; however, patients with malignancy have a particularly elevated risk for developing DVST related to dehydration, chemotherapy effects, and hypercoagulable state [16]. Tumor involvement of the cranium or skull base, dura, or leptomeninges may result in local venous stasis secondary to mass effect on a dural venous sinus, representing an additional mechanism for DVST formation. Specific cancer chemotherapeutic agents that have known association with the development of DVST include asparaginase, cisplatin, and thalidomide [17].

Unenhanced head CT is often the initial imaging modality used to evaluate patients suspected of having DVST, who may present with signs and symptoms of increased intracranial pressure [17]. The classic unenhanced head CT finding in uncomplicated DVST is hyperdensity within the affected dural venous sinus, although this is not invariably present (Fig. 46.2). The superior sagittal and transverse sinuses are most frequently affected in DSVT [16]. The diagnosis is confirmed with CTV or MRV, both of which will depict an intraluminal filling defect within the affected sinus. CTV is faster and usually more accessible after-hours, while MRV is performed alongside MRI of the brain, providing a more sensitive assessment for early venous ischemia or congestion resulting from DVST. In cases of venous infarct, edema or

Fig. 46.2 Dural venous sinus thrombosis. (a) Axial unenhanced head CT image in a patient with lymphoma and headache demonstrates a hyperdense superior sagittal sinus (white arrow), suggesting superior sagittal sinus thrombosis. (b) A more superior unenhanced axial head CT image in the same patient near the vertex demonstrates continued superior sagittal sinus hyperdensity (long arrow) and multiple adjacent hyperdense cortical veins, suggesting thrombosis of these superficial veins as well (short arrow). (c) Postcontrast T1 axial brain MR image in the same patient illustrates a hypointense filling defect in the superior sagittal sinus, compatible with thrombus (white arrow). (d) Gradient echo axial brain MR image demonstrates "blooming" in the superior sagittal sinus, indicative of extracellular blood products (black arrow)



hemorrhage in a non-arterial distribution is common. Treatment of tumor-related DSVT is typically brain irradiation or chemotherapy, depending on tumor histology [10].

Stroke The hypercoagulable state of malignancy constitutes the primary risk factor for the development of cerebral vascular accidents (CVA) in oncologic patients [10]. As noted above, this population is at elevated risk for DVST, which can lead to ischemia and hemorrhagic infarction. Leptomeningeal infiltration of the Virchow-Robin perivascular spaces can result in arterial infarction secondary to thrombosis or vasospasm [10]. Additionally, arterial infarction can occur secondary to herniation as a consequence of compression of large arteries against rigid intracranial structures, such as the cerebral falx or tentorium. For instance, transfalcine herniation can result in anterior cerebral artery (ACA) compression and ipsilateral ACA-territory infarct [1, 8]. Similarly, transtentorial herniation can result in compression of the posterior cerebral artery (PCA) and therefore result in PCA-territory infarct [8]. Less commonly, transalar herniation can result in compression of the carotid terminus and lead to infarction in both the ACA and MCA territories.

Although the initial imaging evaluation of stroke patients regardless of cause is with unenhanced head CT, MRI provides optimal evaluation of tumor location and extent (including leptomeningeal metastatic disease) and is more sensitive in the identification of early ischemia as compared to CT. Areas of acute ischemia are markedly hyperintense on DWI and are typically associated with increased T2 signal. Conventional MRI can provide an overall assessment of vessel caliber and enhancement, but dedicated MRA or MRV sequences provide better detail of arterial and venous pathology.

Spinal Pathology Spinal disease can be divided into pathology involving the osseous spine and that contained within the spinal canal (most commonly epidural or intramedullary involvement). Osseous spine disease is also addressed in the musculoskeletal subsection. Here, we focus primarily on epidural and intramedullary spinal disease, which can both present as oncologic emergencies. Metastatic disease of the spinal epidural space occurs in 5-10% of patients with cancer, most commonly from prostate, breast, and lung primaries [18]. When epidural tumor volume is advanced, it compresses the thecal sac and can result in malignant spinal cord compression (Fig. 46.3). In addition to metastatic disease, spinal cord compression can result from mass effect from primary bone tumors, osseous lymphoma, or multiple myeloma. Spinal epidural metastases localize to the thoracic region in 60% of cases and lumbosacral area in 30% of cases. Cervical spine involvement is less common [18].

Plain radiography is often the initial modality used in the evaluation of spinal cord compression, despite its poor predictive value in determining which patients will have spinal tumor involvement [19]. Vertebral body height loss secondary to pathologic compression is one of the most easily identifiable findings on plain radiography. Compression fractures resulting in loss of vertebral body height \geq 50% is associated with spinal epidural disease in nearly 85% of cases [18]. Erosion of the osseous margins of the spine may be the earliest radiographic sign of intraspinal extension of disease, but this finding is not sensitive [20]. Vertebral pedicle erosion, in particular, may be the most specific finding of epidural disease [18]. CT is superior to plain radiography in accurately depicting bone erosion and assessing compression fractures. A further advantage of CT over plain radiography is that nonosseous disease can often be identified, although MRI is better suited for soft tissue pathology. CT can also be helpful in planning for interventional procedures or surgery.

MRI provides the most sensitive and specific evaluation of bone marrow pathology, epidural tumor and spinal cord compression, and intramedullary spinal disease. Spinal cord compression is discussed in detail in a separate chapter. Metastatic involvement of the spinal cord parenchyma is referred to as intramedullary spinal cord metastasis (ISCM) [18, 21]. Although ISCM has become much more frequently recognized in the era of MRI, spinal epidural disease is still nearly 20 times more common [18]. ISCM affects the cervical, thoracic, and lumbar cords equally and is most often solitary [18]. Bronchogenic carcinoma, particularly small cell carcinoma, accounts for the majority of cases (54%) [18]. MRI with IV contrast is necessary for the diagnosis of ISCM and can effectively discriminate spinal cord edema from enhancing tumor. Intravenous contrast also helps in differentiating tumor with surrounding edema from transverse myelitis.



Fig. 46.3 Postcontrast axial T1 MR image (**a**) and sagittal postcontrast T1 MR image of the thoracic spine (**b**) depict an enhancing mass arising from the T7 vertebra and extending into the epidural space (*white arrows*). There is associated severe cord compression. A single fused

sagittal FDG-PET/CT image of the entire spine (c) demonstrates robust metabolic activity in the tumor at the T7 vertebral level (*white arrow*) and at several other remote vertebral levels (*black arrows*)

Neurologic Complications Additional Acute of Malignancy Neurologic complications of cancer treatment include chemotherapy or brain irradiation-induced brain edema and attendant intracranial hypertension and opportunistic infection from systemic immunosuppression [8]. Infectious meningitis is often undetectable with unenhanced head CT; however, entities such as invasive fungal sinusitis or herpes encephalitis can be apparent on unenhanced head CT and be further characterized with brain MRI. Paraneoplastic limbic encephalitis, which can present with acute-onset confusion, short-term memory loss, hallucinations, and mood changes, is particularly difficult to diagnose clinically as it can be confused with primary psychiatric conditions. Furthermore, limbic encephalitis can have a similar imaging appearance to herpes encephalitis [8]. In 70–80% of patients with limbic encephalitis, MRI FLAIR or T2 sequences show hyperintense signal in one or both medial temporal lobes (Fig. 46.4) [22]. The tumors most frequently implicated are small cell lung cancer, testicular germ-cell neoplasms, thymoma, Hodgkin lymphoma, and teratoma [22].



Fig. 46.4 Single FLAIR axial brain MR image in an emergency department patient with small cell lung cancer and altered mental status depicts bilateral medial temporal hyperintensity (*arrows*). Diagnosis of paraneoplastic limbic encephalitis was based on clinical and imaging characteristics

Chest Emergencies

Central Airway Obstruction Airway compromise necessitating palliative treatment occurs in 20–30% of patients with lung cancer over the course of their lifetime [23, 24]. Airway narrowing can result from intrinsic tracheobronchial disease, extraluminal compression by tumor, or a combination of both [2, 23]. Anatomical distortion of the airways as a result of surgery for lung cancer can also lead to airway compromise [23]. Chest radiography is often the initial imaging test in these patients and may reveal a mass involving the lung parenchyma, hilum, or mediastinum with associated post-obstructive atelectasis or pneumonia [2]. Tracheal deviation or airway narrowing may be present [25].

The imaging gold standard for the assessment of the central airway obstruction is contrast-enhanced CT chest and neck (above the thoracic inlet) (Fig. 46.5) [23]. CT accurately depicts the severity and extent of airway stenosis and helps differentiate intrinsic from extrinsic disease. CT also helps separate primary malignancy from metastases and distinguish tumor from atelectasis or post-obstructive pneumonia [2, 26]. Post-processed images, such as virtual bronchoscopy, can render the tracheobronchial tree in a visual format familiar to the clinician. This can assist in planning palliative interventions, such as stenting or ablative therapies, complementing conventional bronchoscopy to improve the technical success of airway recanalization (Fig. 46.6) [2]. FDG-PET will accurately discriminate a malignant hilar mass from adjacent post-obstructive atelectasis; however, patient preparation, imaging acquisition time, and availability limit its usefulness in the emergent setting [1].

Esophagorespiratory Fistula Esophagorespiratory fistula formation is a relatively rare but potentially devastating complication of esophageal and bronchogenic carcinoma, occurring in up to 22% of esophageal malignancy and around 1% of bronchogenic carcinoma [27, 28]. Nodal metastases and lymphoma can erode into the esophagus and airways and have also been implicated in esophagorespiratory fistula development [28, 29]. The risk of death is related to sepsis from repeated episodes of aspiration or from overwhelming lung infection [27]. Chest radiography findings are nonspecific and can include airspace consolidation, lung abscess, and pleural effusion resulting from aspiration of secretions and ingested material [29].

Fig. 46.5 Patient presenting to the emergency department with dyspnea. This case of advanced ameloblastoma of the right maxillary alveolar ridge in a young man demonstrates tumor mass effect involving the face and upper airway. The lateral scout image from head CT (a) depicts facial deformity as well as tumor replacing much of the nasopharynx (white arrow). Axial contrastenhanced facial CT using soft tissue window (b) and bone window (c) depicts substantial mass effect in the right lower face and widespread local bone destruction. Coronal image (d) demonstrates the complete occlusion of the nasal cavity from tumor invasion and mass effect (short arrow) and substantial mass effect on the oral cavity (long white arrow)



J. Lim et al.



Fluoroscopic evaluation of the aerodigestive tract during the administration of an oral contrast agent will depict simultaneous opacification of the esophagus and tracheobronchial tree distal to the fistula. This procedure is best performed using a low-osmolar, water-soluble, iodinated contrast agent (iohexol), as large volumes of aspirated barium can compromise alveolar oxygen exchange and extravasated barium can incite an inflammatory response in the mediastinal soft tissues. Aspiration of high-osmolar, water-soluble contrast agents can lead to pulmonary edema and should likewise be avoided [30]. Chest CT performed after orally ingested contrast (CT esophagram) may demonstrate a direct communication between the respiratory tract and esophagus at the site of fistula formation with additional findings of ingested contrast agent within the respiratory tract distal to the fistula [29]. Abnormal soft tissue is often identified in the region of the fistula, indicating the site of malignancy [31]. CT is helpful in evaluating the extent and number of fistulas and the presence of a possible esophagorespiratory fistulous communication [29]. Virtual bronchoscopy or esophagography can enhance diagnostic confidence and serve as a component of treatment planning for emergent intervention [29].

Superior Vena Cava Syndrome Malignancy is responsible for 90% of cases of superior vena cava syndrome (SVCS) [32]. Benign causes of SVCS include thrombosis from

indwelling catheters, fibrosing mediastinitis due to immune response to Histoplasma capsulatum or tuberculosis, prior radiotherapy, cardiac pacer wires, and Behçet disease [27, 32, 33]. The clinical features of SVCS relate to venous congestion due to obstruction of the SVC from extrinsic compression by tumor or intraluminal occlusion from bland or tumor thrombus. Primary malignancy involving the superior vena cava resulting in SVCS is exceptionally rare [34]. Because bronchogenic carcinoma accounts for the majority of malignancy-related SVCS, chest radiography may reveal a lung mass with possible associated hilar or mediastinal involvement; however, cross-sectional imaging is necessary to confirm the diagnosis and identify the underlying etiology. Conventional venography was previously the imaging gold standard for diagnosing SVCS, but in contemporary practice, it is employed solely during endovascular intervention [34].

Contrast-enhanced chest CT is currently the preferred imaging modality for the assessment of SVCS. If this condition is suspected on the basis of clinical presentation, CT image acquisition following a 60-second delay optimally opacifies the systemic venous system and ideally characterizes the level (above or below the azygos arch) and extent of SVC obstruction [34]. Practically speaking, however, the diagnosis is often made on routine chest CT or CTPA protocol, which use shorter time delays [34]. Features of contrastenhanced chest CT include complete occlusion of the SVC, an intraluminal filling defect within the SVC, marked narrowing of the SVC from surrounding soft tissue, and opacification of mediastinal or chest wall venous collateral vessels (Fig. 46.7) [34]. Included images of the upper abdomen may demonstrate intense enhancement of the medial segment of the left hepatic lobe ("quadrate lobe"), reflecting collateralization of the superficial epigastric veins and the left portal vein [34]. CT imaging findings of SVC obstruction may precede the development of the clinically apparent syndrome 597 rention [32]. The

and present an opportunity for early intervention [32]. The underlying cause of the obstruction, most commonly bronchogenic carcinoma, lymphoma, or extrathoracic metastatic disease when considering malignant etiologies, will also be depicted at CT, representing a critical diagnostic advantage over catheter venography [31]. Although enhancement of soft tissue within the SVC represents intraluminal tumor, presumptive soft tissue enhancement without pre-contrast images for reference can be misleading. CT axial images and post-processed coronal and sagittal reformatted images also assist in planning for endovascular interventions. MRI approaches 100% sensitivity and specificity in the diagnosis of SVSC but is rarely used in the emergent setting due to limited scanner availability and patient factors [32].

Endovascular stenting has replaced chemotherapy and radiation therapy as the treatment of first resort in malignant SVCS, except in the case of chemotherapy-sensitive lymphoma, due to rapid clinical response and established longterm patency [32]. SVC stenting is carried out in the angiography suite. Under ultrasound guidance, the internal jugular, femoral, or subclavian vein is accessed, and superior vena cavography is performed to confirm the extent of disease and inform stent selection and placement. Using guidewires, the obstruction is traversed, and progressive dilatation of the obstructed lumen is optionally preformed prior to stent placement. In addition, local thrombolysis and mechanical thrombectomy may reduce the length of obstruction and risk of pulmonary emboli; however, thrombolysis has been associated with an increased risk of periprocedural bleeding [32]. A self-expanding endoprosthesis is deployed across the obstruction, taking care to not apply excessive pressure, which can result in SVC rupture and cardiac tamponade [32]. A chest radiograph is usually obtained following the procedure to confirm satisfactory stent placement and to serve as a baseline reference for future imaging. Reduction in symptoms is immediate, and clinical response to stenting is around 96% in bronchogenic carcinoma [35]. Recurrence



Fig. 46.7 Axial (**a**) and coronal (**b**) contrast-enhanced chest CT images demonstrate malignant SVC occlusion in this patient presenting with clinical SVC syndrome. Bulky enhancing soft tissue mass (*short arrows*) obliterates the expected location of the SVC. The inferior por-

tion of the SVC is narrowed and displaced medially on the coronal image (b), superior to which SVC cutoff from tumor involvement is noted (*arrowhead*). Tissue sampling revealed bronchogenic carcinoma. A right pleural effusion is also present

of obstruction has been reported with an incidence of 0-40% [36]. Repeat stent placement is indicated in these cases and is associated with a high success rate [36].

Massive Hemoptysis Massive hemoptysis is generally defined as expectoration of \geq 300–600 mL of blood within a 24-hour period and is associated with a 9-38% mortality rate [37, 38]. While pulmonary tuberculosis is the leading cause of hemoptysis worldwide, bronchogenic carcinoma is the most common malignant etiology [38]. Unstable patients presenting with massive hemoptysis are usually initially managed with bronchoscopy, which localizes the site of bleeding in the airways with a 73-93% diagnostic yield, and can be used for hemostasis [37]. Chest CTA performs equally well in determining the site of hemorrhage and is superior to bronchoscopy in identifying the underlying cause [37]. Chest radiography may reveal a lung mass, cavitary lesion, consolidation, or mediastinal mass, but up to 10% of patients with malignancy as the cause for hemoptysis may have a normal chest radiograph [39]. In most patients, chest CTA with postprocessing is the diagnostic study of choice and can identify the site, underlying cause, and vascular origin of bleeding with a high degree of accuracy [38]. In 90% of patients with massive hemoptysis, a bronchial artery source is implicated [37]. On chest CTA, an abnormally dilated (≥ 2 mm diameter) or tortuous bronchial artery is suspicious for source of bleeding and targeted for embolization (Fig. 46.8) [38]. Active extravasation of contrast, while highly specific, is relatively rare and presents in only 3.6–10.9% of cases [38].

In addition to high diagnostic yield, chest CTA aids in planning the approach for catheter angiography and has been shown to decrease procedure time and technical success rate of subsequent embolotherapy [37, 38]. Bronchial artery embolization is an effective and safe treatment for massive hemoptysis with documented 73–99% success in effecting immediate control of bleeding [38]. Recurrence occurs in

J. Lim et al.

10–29% in the first month and relates to incomplete embolization due to extensive disease or an occult non-bronchial source [38]. These patients are usually effectively retreated with embolization.

Pulmonary Embolism Acute pulmonary embolism (PE) is a leading diagnostic consideration in the oncologic patient presenting with acute chest pain. Left untreated, acute PE can be fatal [40]. Cancer patients are at particularly heightened risk for thromboembolic disease due to hypercoagulability. Used in conjunction with various clinical decision instruments to measure pretest probability, imaging plays an essential role in the diagnosis. As with any patient presenting with acute chest pain, chest radiography is used as an initial screening modality, although most patients with acute PE will have a normal chest radiograph [40]. The Westermark sign (geographic lucency related to regional oligemia from pulmonary arterial obstruction) and Hampton hump (wedgeshaped, peripheral consolidation representing infarcted lung tissue), though classically associated with acute PE, are rarely seen in practice. More commonly, atelectasis or airspace consolidation may be present in a minority of patients, but these findings are not specific [40]. The chief utility of chest radiography is to exclude other etiologies for chest pain, including pneumonia, pneumothorax, or pleural effusion [40]. In addition, chest radiography is helpful in interpreting ventilation-perfusion scintigraphy.

After excluding other causes of acute chest pain, CTPA is the imaging modality of first resort for diagnosis of acute PE. CTPA has well-established diagnostic accuracy, with 86% positive predictive value and 95% negative predictive value, having surpassed catheter pulmonary arteriography as the imaging gold standard [40, 41]. On CTPA, a centrally located hypodense intraluminal filling defect within the densely opacified pulmonary arterial system is seen in acute PE (Fig. 46.9) [40, 41]. Suboptimal vascular opacifi-

Fig. 46.8 (a) Coronal contrast-enhanced chest CT in a patient with massive hemoptysis depicts a hypodense right suprahilar mass (black asterisk) and subtotal right lung collapse (arrow). Pleural fluid is also seen adjacent to the collapsed lung (white asterisk). (b) Accompanying bronchial artery angiography with dilated and tortuous bronchial arteries, which were subsequently embolized, resulting in hemostasis





Fig. 46.9 Coronal (**a**) and axial (**b**) contrast-enhanced chest CT images demonstrate a hypodense filling defect in the contrast-opacified lumen of the right main pulmonary artery, with extension into the interlobar

pulmonary artery, compatible with acute pulmonary embolism (*white arrows*). Also visible in the upper abdomen are multiple hepatic metastatic lesions (*black arrows*) and malignant ascites (*asterisk*)

cation and respiratory motion can reduce the diagnostic accuracy of CTPA, with a nondiagnostic rate of around 6%, whereas V/Q scintigraphy is inconclusive in over 25% of cases [40]. Incorporation of dual energy in CTPA has demonstrated improved sensitivity in detecting segmental and subsegmental pulmonary emboli [42]. Incidental but important additional findings can be detected with CTPA, such as pulmonary nodules and mediastinal lymphadenop-athy [41].

Doppler and grayscale sonography of the deep venous system in the extremities is frequently employed as an adjunct to CTPA, since 36-45% of patients with acute PE have proximal deep venous thrombosis (DVT) [41]. Features of DVT on sonography include visualization of intraluminal thrombus, loss of venous compressibility, venous distention, and absence or diminution of Doppler color or spectral signal [40]. Serial negative extremity ultrasound examinations may obviate the need for additional investigative procedures or treatment in patients who are not candidates for CTPA related to renal insufficiency or adverse reaction to iodinated contrast or for whom CTPA or V/Q imaging is indeterminate [40]. By the same token, patients with evidence of extremity DVT by ultrasound and high pretest clinical suspicion for acute PE can be treated empirically without confirmatory imaging [41]. MR pulmonary angiography (MRPA) with and without gadolinium-based contrast agents has been studied in the evaluation of acute PE. Despite good documented diagnostic performance in technically adequate examinations, MRPA is currently not recommended for routine use due to limited availability and expertise in interpretation and suboptimal technical success [41].

Pericardial Effusion and Pericardial Tamponade Pericardial effusion and tamponade can be caused by primary malignancy in the chest, metastatic disease from a remote primary location, and following treatment for malignancy [43]. Common primary malignancies associated with peri-

cardial effusion include bronchogenic carcinoma, breast carcinoma, lymphoma, and leukemia [43]. Lymphatic obstruction by tumor deposits is the predominant mechanism responsible for the development of malignant pericardial effusion, although direct contiguous extension (bronchogenic, esophageal, and breast) and hematogenous spread of tumor (lymphoma and leukemia) are other notable pathways for pericardial malignant disease [44]. Primary pericardial malignancies, such as mesothelioma or fibrosarcoma, are exceptionally rare etiologies for pericardial effusion [45, 46]. If accumulation of pericardial fluid is rapid, as little as 200– 250 mL can result in tamponade physiology due to the relative inextensibility of the parietal pericardium [44].

A large pericardial effusion will result in an enlarged cardiac silhouette with a "water bottle" conformation on chest radiography [47]. On echocardiography, a pericardial effusion is readily apparent, and prolonged right atrial collapse during late diastole and right ventricular collapse in early diastole are characteristic for cardiac tamponade [47]. In the emergent setting, contrast-enhanced chest CT is ideally suited to establish the presence, extent, and possible cause of a pericardial effusion leading to tamponade. Fluid measuring near-water attenuation suggests a simple effusion, whereas higher-density fluid caused by hemorrhagic tumor deposits and cellular debris is often seen in malignant pericardial effusion [48]. An irregularly thickened (≥ 2 mm) pericardial lining with enhancing nodularity is highly suspicious for a malignant etiology in the appropriate setting, although mycobacterial infection can also have this appearance (Fig. 46.10) [48]. If a lung or chest wall mass and/or mediastinal adenopathy are present on CT, these findings would lend support to a malignant cause of a pericardial effusion. Definitive diagnosis is made by cytological examination of aspirated fluid. Treatment by pericardiocentesis with or without drain placement is performed with CT or echocardiography guidance [25, 31].

а

Fig. 46.10 Contrastenhanced axial chest CT in a patient with metastatic melanoma depicts a large pericardial effusion (P). Findings are compatible with malignant effusion by CT given the effusion density and the nodular enhancing tumor foci (*long arrow*). Flattening of the right ventricular wall is suggestive of cardiac tamponade physiology (*short arrows*)



Fig. 46.11 Certain tumors, particularly late in a patient's disease course, can exert substantial mass effect in the peritoneal cavity. By virtue of this mass effect, they can compress the GI or GU tracts and result in early satiety, nausea, vomiting, and bowel or ureteral obstruction, respectively. A contrast-enhanced axial CT image of the abdomen and pelvis (**a**) demonstrates multiple large peripherally enhancing peri-

Abdominopelvic Emergencies

Bowel Emergencies Bowel obstruction in patients with abdominal or pelvic malignancy results from impingent on the bowel lumen, either from intrinsic mural disease or from extramural compression usually from serosal implants (Fig. 46.11). In addition, many oncologic patients have undergone prior abdominal or pelvic surgery, which predisposes to bowel obstruction secondary to adhesions [49]. Bowel obstruction may also result from complication of treatment. For example, nivolumab, which is an immunotherapeutic agent for lung cancer, has been associated with increased risk of small bowel obstruction and perforation [50]. Colorectal carcinoma (10–30%) and ovarian malignancy (20-50%) are the most common cancers associated with bowel obstruction. Metastatic disease is more commonly implicated than primary malignancy in small bowel obstruction, as it is usually secondary to compressive effects

toneal implants from choriocarcinoma (representative lesion indicated by a black asterisk), which significantly displace normal structures. In a different patient, axial (**b**) and sagittal (**c**) contrast-enhanced CT images of the abdomen and pelvis demonstrate bulky multicystic masses (*white asterisks*) with very thin walls arising from a mucinous ovarian cystadenoma

from peritoneal metastases; however, primary gastrointestinal malignancy can rarely result in obstruction (Fig. 46.12). Additional primary cancers that can present with peritoneal metastases, and predispose to small bowel obstruction, include gastric, pancreatic, breast, and lung [49].

In the emergent setting, suspected bowel obstruction is often initially assessed with abdominal radiography. Supine abdominal radiographs may demonstrate abnormally dilated small bowel (≥ 2.5 cm in diameter) in small bowel obstruction and small bowel and colon in colonic obstruction [51]. Upright abdominal radiographs may show multiple air-fluid levels or, in the case of viscus perforation, free air accumulation under the diaphragms. Plain radiography is only around 50–60% accurate in the evaluation of small bowel obstruction and does not adequately predict the site of obstruction [51, 52]. Contrast-enhanced CT of the abdomen and pelvis is the imaging of choice when evaluating for bowel obstruc-



Fig. 46.12 A 57-year-old presented to the emergency department with nausea and vomiting. Sagittal (**a**), axial (**b**), and coronal (**c**) contrastenhanced CT images of the abdomen and pelvis demonstrate a dilated fluid-filled stomach (*asterisk*). At the gastric antrum, there is irregular,

enhancing wall thickening resulting in gastric outlet obstruction (*arrows*). Pathology revealed poorly differentiated gastric adenocarcinoma of the distal stomach and pylorus

tion, with far superior accuracy in securing the diagnosis (95%) and locating the site of obstruction [52]. On CT, dilated small bowel ≥ 2.5 cm (outer wall-to-outer wall) is seen proximal to a discrete transition point, distal to which the remainder of the small bowel and the colon are relatively decompressed (Fig. 46.13) [51]. Abnormal enhancing or necrotic soft tissue may be seen in the region of the transition point, representing peritoneal metastatic disease or, rarely, primary bowel malignancy. In contradistinction to small bowel obstruction, large bowel obstruction is most often caused by primary colorectal malignancy as opposed to metastatic disease [53]. Large bowel obstruction is diagnosed when both small bowel and colon (≥ 9 cm for cecum, otherwise ≥ 6 cm) are dilated proximal to a transition point within the colon [54]. Differentiating a transition point from a postinflammatory or ischemic stricture in large bowel obstruction on CT can be difficult given similar imaging appearances. The presence of enlarged local lymph nodes may suggest a malignant etiology. Oral or rectal contrast is typically not administered except for suspected bowel complications, such as perforation or fistula or sinus tract formation [52, 55]. MRI is generally not used for evaluating bowel obstruction and is often reserved for pregnant or pediatric patients, in whom ionization radiation exposure is of particular concern.

Among patients with spontaneous pneumoperitoneum secondary to bowel perforation, 14% are attributable to malignancy [56]. Plain radiography is diagnostic in only 30–59% of cases of free intraperitoneal gas but approaches 100% accuracy in large-volume pneumoperitoneum [56]. Radiographic signs include free air under the diaphragms, increased bowel wall visualization from the presence of extraluminal and intraluminal gas (Rigler sign), lucency out-



Fig. 46.13 Single coronal contrast-enhanced CT image of the abdomen and pelvis in a patient presenting with nausea and vomiting demonstrates a heterogeneously enhancing cecal-region mass (*arrow*). There are dilated, fluid-filled segments of small bowel, compatible with small bowel obstruction. Pathology revealed appendiceal carcinoma. Note serosal metastatic implants (*arrowhead*)

lining the falciform ligament (falciform ligament sign), and air outlining the entire abdominal cavity (football sign) and a hyperlucent liver [56]. Abdominal radiography cannot predict the site of perforation. CT, however, is exceedingly sensitive for the detection of free intraperitoneal gas (96–100%) and can correctly identify the site of perforation with 80–90% accuracy [56]. CT features of bowel perforation include discrete bowel wall defect and extraluminal gas, oral or rectal contrast, or bowel contents [56]. Both colorectal carcinoma and gastrointestinal lymphoma are complicated by perforation with a prevalence of up to 9%, and systemic chemotherapy has been observed to further increase the risk of perforation in bowel lymphoma [57].

Ischemic colitis can coexist with colonic malignancy with an incidence of up to 7%, and 20% of patients with ischemic colitis have an underlying colorectal cancer [58, 59]. Ischemic colitis secondary to malignancy can result from increased intraluminal pressure and subsequent diminished blood flow in the dilated colon proximal to site of primary malignancy. In cases without mechanical obstruction, bacterial overgrowth in stagnant segments of colon has been implicated in ischemic colitis [60]. Pancreatic adenocarcinoma can compress the superior mesenteric artery and vein, resulting in critically diminished perfusion and subsequent bowel ischemia [60]. Metastases to the mesenteric root, including colon, breast, ovarian, and lung primaries, can similarly result in bowel obstruction by direct vascular impingement [1]. In ischemic bowel, contrast-enhanced CT of the abdomen and pelvis is the imaging modality of choice, and features include mural thickening, submucosal edema (low attenuation) or hemorrhage (high attenuation), engorgement of mesenteric vessels, mesenteric edema, altered or absent mucosal enhancement, and intramural gas or portal venous gas. Hypoenhancement of the bowel wall is considered the most specific sign for ischemic bowel, although these findings are often subtle. Dual-energy CT can increase the conspicuity of bowel wall hypoenhancement through the use of iodine maps or overlay images [61]. Permeation of intraluminal gas across damaged mucosa causes intramural gas and appears as focal or circumferential locules of air within the colonic wall (pneumatosis coli). Gas can then propagate into the mesenteric and portal veins, giving the appearance of air-attenuation filling defects within these vessels.

Intussusception is characterized by telescoping of a segment of bowel along with its corresponding mesentery into an adjacent segment of bowel. Whereas 95% of cases of intussusception are idiopathic in children, 80–90% of adult cases are associated with underlying mass lesions, referred to as "lead points," and are most commonly polypoid bowel neoplasms [62, 63]. In 30% of small bowel and 60% of colonic intussusceptions, an intramural or extrinsic lead point will be malignant [62]. With CT, intussusception is characterized by a "bowel-within-bowel" appearance with direct visualization of an inner bowel segment and surrounding fat (intussusceptum) enveloped by an adjacent outer segment of bowel (intussuscipiens), with or without accompanying vessels, rendering a targetoid or sausage-like appearance [53, 63]. Intussusception can be complicated by bowel obstruction or ischemia, the features of which are described above.

Spontaneous Intra-abdominal Hemorrhage Spontaneous hemorrhage from visceral organ malignancy is a rare, but potentially catastrophic, oncologic emergency. Around 10-15% of patients with hepatocellular carcinoma will present to the emergency department with tumor rupture leading to intraparenchymal hematoma, subcapsular hematoma, hemoperitoneum, or some combination. Risk factors for spontaneous hemorrhage of hepatocellular carcinoma include peripheral or subcapsular location and large tumor size [64]. Hypervascular metastases to the liver, such as melanoma and renal cell carcinoma, are at particularly heightened risk for spontaneous hemorrhage. Reports of spontaneous hemorrhage from primary angiosarcoma of the liver have been documented [3]. Although spontaneous splenic rupture is more commonly reported in infection, notably cytomegalovirus. Epstein-Barr virus, or malaria, lymphomatous or leukemic splenic involvement can also give rise to severe hemorrhage, necessitating immediate endovascular treatment or splenectomy [65].

Spontaneous tumor rupture is evaluated with contrastenhanced CT of the abdomen and pelvis in order to confirm the presence of rupture, identify the site of involvement, and determine the extent of hemorrhage (Fig. 46.14). The appearance of extravascular blood on CT varies in a time-dependent manner. Acute blood products will demonstrate an attenuation in the range of 30-45 HU, whereas clotted blood has attenuation values from 45 to 70 HU [3]. When a focal area of high-density clotted blood is seen in the abdomen or pelvis on a background of lower-density unclotted blood, it is referred to as the "sentinel clot" and points to the site of primary source of bleeding [3, 53]. On occasion, ongoing bleeding can be seen during the acquisition of CT images as active extravasation of contrast, with focal areas of highdensity (85-370 HU) extraluminal IV contrast material and surrounding clotted hematoma [66]. Ongoing bleeding requires emergent surgical or endovascular interventions to control blood loss. In this setting, particularly with unstable patients, catheter angiography and catheter-directed embolization can aid in management.

Urinary Obstruction Approximately one-quarter of patients with pelvic or retroperitoneal malignancy will develop life-threatening urinary obstruction [1]. Progressive urinary tract obstruction most often results from compression or invasion of the ureters, most commonly the distal one-third of the ureters below the level of the common iliac



Fig. 46.14 A 47-year-old patient presented to the emergency department with acute-onset right upper quadrant pain. Coronal (**a**), axial (**b**), and sagittal (**c**) contrast-enhanced CT images of the abdomen reveal a heterogeneously enhancing exophytic hepatic mass (*asterisk*) arising

from the right hepatic lobe. There is surrounding high-density fluid, compatible with blood products (*arrows*). Pathology revealed hemorrhagic hepatocellular carcinoma

arteries, by primary malignancies such as prostate, bladder, cervical, ovarian, or colorectal [1, 67]. Lymphoma and sarcoma, and less likely retroperitoneal metastatic disease from primary cancers such as cervix, urinary bladder, prostate, colorectal, ovary, and testes, can likewise result in ureteric obstruction by direct tumor invasion or compression [1, 67].

Hydronephrosis and hydroureter can be identified by ultrasound as expansion of the renal collecting systems and ureters with anechoic urine. The obstructing mass may not be evident on ultrasound, and advanced cross-sectional imaging is usually required to assess for the presence, size, and location of the malignancy. In the emergent setting, CT is often initially employed either as a standard single-phase postcontrast CT or formal CT urography. In single-phase CT, both the obstructed urinary tract and offending lesion are clearly depicted. A delayed enhancement pattern of the ipsilateral kidney may be present, reflecting compromised excretion of contrast due to downstream obstruction (Fig. 46.15). CT urography is not often employed in the acute setting but would consist of an unenhanced image set, followed by nephrogenic phase (100 seconds after IV contrast administration) and excretory phase (3 minutes after IV contrast administration) imaging [68]. The urographic component, however, may be limited in severe obstruction, which would inhibit excretion of contrast. Treatment of malignant urinary obstruction is palliative, as median survival is measured in months [67, 69]. Nephroureteral stent and percutaneous nephrostomy tube placement are the most common interventions for palliative urinary diversion [67].

Biliary Obstruction Obstruction of the biliary system from primary tumor invasion or compression of the bile ducts by hilar nodal metastases can result in significant morbidity and mortality [70]. Pancreatic adenocarcinoma, periampullary

malignancy, and cholangiocarcinoma are most the most commonly implicated primary malignancies associated with biliary obstruction [70, 71]. In patients with known or suspecting malignancy presenting with jaundice, abdominal pain, and/or laboratory evidence of biliary obstruction, the obstructing mass and associated obstruction can be confirmed by abdominal ultrasound, CT, or MRI. CT and MRI offer the additional advantage of preprocedural planning and staging [72]. MR cholangiopancreatography (MRCP), which is always acquired in combination with conventional MRI of the abdomen, is a noninvasive imaging technique that provides exquisite anatomic detail of the dilated biliary tree and can often exclude choledocholithiasis as a cause of obstruction [73]. Endoscopic ultrasound can be complementary to CT and MRI/MRCP in difficult cases [72]. Pancreatic adenocarcinoma is typically characterized as an ill-defined and infiltrative mass, which is hypoattenuating with respect to normal pancreatic parenchyma on CT. In rare cases, pancreatic adenocarcinoma is indistinguishable from normal pancreatic tissue. When this occurs, the presence of malignancy in the pancreatic head can be inferred by the "double duct" sign, representing simultaneous common biliary and main pancreatic ductal dilatation [70]. Cholangiocarcinoma is classified as intrahepatic, hilar, or extrahepatic, and most occur at the bifurcation of the hepatic ducts (Klatskin tumor) [70]. At the time of presentation, the common imaging appearance of cholangiocarcinoma is biliary ductal dilatation, which can be focal or diffuse. On cross-sectional imaging, infiltrative hilar cholangiocarcinoma is further characterized by an ill-defined soft tissue mass with associated by bile wall thickening or complete duct luminal obliteration (Fig. 46.16). Intrahepatic (mass-forming) cholangiocarcinoma is hypovascular and demonstrates gradual centripetal enhancement in a time-dependent manner following IV contrast administration [70].



Fig. 46.15 A 70-year-old female with urinary bladder urothelial carcinoma. (a) Initial sonographic image shows dilation of the renal collecting system with calyceal blunting (*short white arrows*). Subsequent sonographic image of the pelvis (b) shows dilation of the distal ureter (long white arrow), with vascularized soft tissue mass in the urinary bladder. Coronal CT image (c) redemonstrates hydronephrosis (*short*)

black arrow) and confirms the enhancing mass in the right lateral aspect of the urinary bladder (*long black arrow*). Also present is relative hypoenhancement of the right kidney compared to the left kidney, termed a "delayed nephrogram," a finding often seen with higher-grade ureteral obstruction



Fig. 46.16 Contrast-enhanced axial CT image of the abdomen (**a**) and transverse sonographic image centered at the hepatic hilum in a patient presenting to the ED with jaundice and right upper quadrant pain. The CT image depicts intrahepatic biliary ductal dilation (short arrow) and a hypoenhancing hepatic hilar mass, typical of cholangiocarcinoma

(Klatskin tumor). Sonographic image of the right upper abdominal quadrant (**b**) demonstrates biliary ductal dilation and an ill-defined heterogeneous mass with Doppler signal, indicating vascularity. RHD, right hepatic duct; LHD, left hepatic duct

In patients with advanced malignant biliary obstruction, palliative biliary diversion can be performed surgically, endoscopically, or via a percutaneous approach. A surgical approach results in significant reduction in recurrent obstruction but is associated with a higher complication rate [71]. Endoscopic and percutaneous stenting using both plastic and metallic stents is associated with lower complication rates but higher reocclusion rates and need for repeat procedures [71, 74]. In general practice, bypassing distal biliary obstruction is usually initially attempted with ERCP, and percutaneous biliary drainage and stenting are reserved for endoscopic technical failure [74]. For proximal obstruction, both approaches can be taken, sometimes in combination; however, a percutaneous approach, which is usually carried out in the interventional radiology suite, may be more appropriate when drainage of segmentally dilated bile ducts is desired. This is because percutaneous ultrasound can first identify these structures, allowing for a more targeted subsequent intervention using fluoroscopic guidance [74].

Musculoskeletal Emergencies

Pathologic Fractures The musculoskeletal system is comprised of the axial and appendicular skeleton as well as the supporting muscles and soft tissues. Oncologic emergencies can affect both structural components, but skeletal complications are much more frequently encountered. Bone is the most common site for metastatic disease [75]. At postmortem examination, the incidence of metastatic bone disease was 73% in breast cancer, followed in frequency by prostate (68%), thyroid (42%), lung (36%), and renal (35%) [75]. The burden of osseous metastatic disease correlates to the frequency of skeletal-related events (SREs), which include fractures, surgical or therapeutic intervention for bone lesions, spinal cord compression, and hypercalcemia of malignancy [76].

The destruction of trabecular and cortical bone by primary or metastatic tumor degrades the intrinsic structural stability, and therefore the weight-bearing capabilities of bone, predisposing patients to pathologic fractures. These fractures occur in both the axial and appendicular skeleton but are more debilitating in the vertebra, pelvis, or lower extremities (Fig. 46.17).

Appendicular Skeleton Regarding the long bones, plain radiographs provide rapid initial screening. When fractured, subsequent orthopedic consultation and stabilization of a pathologic fracture restores functionality and decreases pain. If plain radiographs are not revealing, but there remains a high clinical concern for fracture, further imaging may be warranted. Specifically, in patients with low bone mineral density, nondisplaced fractures can be occult by plain radiography, and if the patient cannot bear weight or has unexplained pain, further evaluation with cross-sectional imaging may be warranted. Unenhanced MRI provides very sensitive evaluation for both metastatic involvement of bone and acute fracture; however, after-hours availability is not widespread. Normal bone marrow varies by location within bone and



Fig. 46.17 AP radiograph of the humerus demonstrates a geographic, moth-eaten, lytic, destructive lesion, with cortical endosteal thinning and mild marrow cavity expansion (**a**). (**b**) Magnified image of the associated pathologic fracture with adjacent cortical fragments (*arrows*). Coronal T1 postcontrast fat-saturated MR image (**c**) demonstrates

enhancing soft tissue mass replacing the normal fatty marrow cavity (*short arrow*) and obliterating the T1-hypointense cortex (normal cortex shown with *long arrow*). Diagnosis is plasmacytoma with pathologic fracture

patient age; however, in general normal marrow can be categorized as hematopoietic or fatty. Fatty marrow is T1-hyperintense and T2-hyperintense, while hematopoietic marrow is slightly hypointense on both sequences. The proportion of hematopoietic marrow decreases with patient age. Marrow can be replaced or infiltrated by metastases, leukemia, or lymphoma, resulting in T1 signal that is hypointense relative to skeletal muscle (which serves as an internal frame of reference) [77]. On fat-saturated T2 and STIR imaging, malignancy is usually hyperintense, resulting from intrinsically higher water content and reactive edema [77]. In the case of fracture, fluid-sensitive MRI sequences (T2 and T2 or PD with fat saturation) will demonstrate robust marrow edema, while T1 sequences are used to localize the site of cortical disruption. Unenhanced CT can be used in the emergent setting to identify and characterize fractures. The primary limitations of unenhanced CT in the evaluation of malignant involvement include diminished sensitivity for malignant soft tissue and early marrow metastatic disease.

Vertebral Fractures Metastatic disease to the spine is the most common form of skeletal malignancy, and osteolytic bone destruction may occur in up to two-thirds of patients [75]. As previously noted, plain radiography is widely available and rapid but relatively insensitive for pathologic vertebral body fractures. In the setting of trauma in cancer patients, if there is clinical concern for fracture, unenhanced CT is the imaging modality of choice for identification and characterization of fractures. With MRI, the presence of bone marrow edema (T2 or STIR hyperintensity) indicates an acute time course. The addition of IV contrast to an MR examination provides a more sensitive evaluation of epidural tumor extension but will not further characterize the underlying osseous malignancy and is not necessary in diagnosing fracture. This is because bone metastases do not always enhance after the administration of gadolinium. Sclerotic metastatic disease and treated or partially treated tumor foci, for example, typically do not enhance but will still be detected on MRI without IV contrast [77]. CT-guided vertebroplasty can help reduce pain in a wide variety of osteolytic vertebral lesions [78].

In the setting of vertebral fracture, spinal cord compression can result from retropulsion of bone fragments or soft tissue tumor into the spinal canal. Additionally, local hemorrhage resulting from the fracture can cause local mass effect and exacerbate spinal canal narrowing. MRI provides the optimal assessment of the spinal cord, including the degree and cause of narrowing and the presence or absence of spinal cord edema. For further information on this topic, please see the dedicated chapter on spinal cord compression. Imageguided treatment of this entity is discussed in the section on neuroradiologic emergencies. Impending Fractures The Mirels classification system provides an overview of the risk of impending fracture in long bones, with a score ≥ 8 suggesting prophylactic surgical fixation [79–81]. The calculation of the Mirels score incorporates lesion site, nature of the osseous lesion, lesion size relative to cortical thickness, and pain, as follows:

- Location: upper limb (1 point), lower limb (2 points), and trochanteric region in the proximal femur (3 points).
- Lesion: blastic (1 point), mixed (2 points), and lytic (3 points).
- Size: the size of lesion expressed as a proportion of cortical involvement less than 1/3 (1 point), 1/3 to 2/3 (2 points), and greater than 2/3 (3 points) [79, 81].
- Pain: subjectively classified as mild (1 point), moderate (2 points), or functional (3 points).

Imaging, usually consisting of both CT and plain radiography, is necessary in computing a Mirels score and can guide patient management by suggesting fixation prior to fracture.

Case Study

Introduction: A 67-year-old woman with 40-pack-year smoking history presents with a 1-month history of worsening lower back pain. She also reports that she has had 60-pound weight loss since the preceding year due to poor appetite and low energy from coughing all night. Physical exam is positive for diminished breath sounds over the right chest and tenderness to palpation over the mid-lower back. She undergoes chest radiography and CT of the abdomen and pelvis (Fig. 46.18). She returns to the emergency department 1 year later with acutely worsening low back pain and bilateral lower extremity hyperreflexia (Figs. 46.19 and 46.20).

Discussion: Findings are compatible with pathologic thoracic vertebral fracture secondary to metastatic disease from lung cancer. The spine is the most common location for skeletal metastasis and occurs in up to 10% of newly diagnosed cancers. Metastasis to the spine occurs due to hematogenous dissemination of malignant cells. On radiography and CT, metastases are classified as osteoblastic, mixed or osteolytic. Common primary malignancies that metastasize to the spine are lung (osteolytic), breast (mixed), and prostate (osteoblastic). In this case, the patient presented with osteolytic metastasis from metastatic lung cancer. In addition to metastasis, the differential diagnosis for lytic lesions in the spine includes primary bone tumors, multiple myeloma, lymphoma, discitis/osteomyelitis, and metabolic disorders.

Patients with metastatic disease to the spine present with spinal tenderness and, in more advanced cases, neurologic deficits. This may or may not be seen in setting of pathologic



Fig. 46.18 *Case Study.* Imaging at initial patient encounter. Frontal (**a**) and lateral (**b**) chest radiographs demonstrate a lobulated right hilar/ right upper lobe mass (*arrows* in images **a** and **b**) highly suggestive of

malignancy. (c) Sagittal CT image of the abdomen demonstrates subtle erosion of the posterior vertebral body cortex of T12 (*arrow*)



Fig. 46.19 *Case Study.* Imaging at follow-up patient encounter, 1 year later. Sagittal CT of the chest (**a**) demonstrates a wedging compression fracture of the T12 vertebral body with 50% anterior height loss (*arrow*); however, the age of this fracture is not definitively established

by CT. T2-weighted MRI with fat saturation (**b**) demonstrates an acute pathologic compression fracture of T12 with bright marrow signal confirming the acute time course of this fracture

fracture and spinal cord compression. A Mirels score can be used to assess the risk of pathologic fracture, with a score greater or equal to 9 suggesting prophylactic fixation be performed. Treatment of pathologic fracture is multidisciplinary and involves a combination of chemotherapy, steroids, surgery, and radiation. There are three key factors involved in determining treatment: neurologic deficit, patient factors, and spinal stability. A spinal instability neoplastic score is often used evaluate for spinal stability, with a score of 7 to 18 warranting surgical consultation.

Pain Bone metastases are the most common cause of cancer-related pain, although the majority of individual metastatic bone lesions are not painful [75]. Radiographs

can identify lytic bone lesions as an area of radiolucency but only after a loss of 30% or more of bone mineral density; radiographs can be used as a screening tool, but continued clinical concern for a metastatic lesion in the presence of normal radiographs may warrant cross-sectional imaging. Unenhanced CT provides detailed bone anatomy and can be useful in the assessment of impending fracture risk or in preoperative planning. MRI provides very sensitive evaluation, identifying smaller nondestructive regions of marrow infiltration or metastatic disease [77]. Image-guided percutaneous cryoablation has proven effective for bone pain management. Radiofrequency ablation is an additional treatment option but may cause a temporary pain exacerbation.



Fig. 46.20 *Case Study.* Imaging at follow-up patient encounter, 1 year later (**a**–**c**). Bright hot iron (*yellow/white*) color in axial fused FDG-PET/CT image indicates increased glucose (FDG) uptake in the highly metabolically active primary right hilar/right upper lobe malignancy. Bright hot iron color in (**b**) indicates metastatic lesion in the T12 vertebral body. Full-body three-dimensional FDG-PET projection in (**c**)

demonstrates the primary tumor as a cluster of black masses (*long arrow*), vertebral metastasis (*short arrow*), and physiologic uptake in the renal collecting systems and urinary bladder due to excretion of FDG (*stars*). Metastatic right supraclavicular lymphadenopathy is demonstrated superior to the primary tumor

Hypercalcemia Hypercalcemia of malignancy comprises more than one-third of all cases of hypercalcemia presenting to the emergency department [47]. As many as one-third of cancer patients will experience hypercalcemia at some point in their disease course [25]. Hypercalcemia of malignancy can result from tumoral or systemic release of parathyroid hormone-related peptide (PTHrP), osseous metastases resulting in direct osteoclastic stimulation and osteolysis, and secretion of vitamin D analogues by the tumor [25, 47]. Radiologic manifestations of hypercalcemia of malignancy can include osteopenia and osteoporosis, resorption of bone at the distal clavicle and about the sacroiliac joints, and nephrocalcinosis and nephrolithiasis.

References

- Katabathina VS, Restrepo CS, Betancourt Cuellar SL, Riascos RF, Menias CO. Imaging of oncologic emergencies: what every radiologist should know. Radiographics. 2013;33(6):1533–53.
- Nair A, Godoy MC, Holden EL, Madden BP, Chua F, Ost DE, et al. Multidetector CT and postprocessing in planning and assisting in minimally invasive bronchoscopic airway interventions. Radiographics. 2012;32(5):E201–32.
- Lubner M, Menias C, Rucker C, Bhalla S, Peterson CM, Wang L, et al. Blood in the belly: CT findings of hemoperitoneum. Radiographics. 2007;27(1):109–25.
- Ostby SA, Evans JG, Smith HJ, Boitano TKL, Toboni MD, Heimann MA, et al. Reducing emergency department (ED) computed tomography (CT) utilization in women treated for gynecologic cancers. Gynecol Oncol. 2020;156(2):288–92.

- Murray N, Darras KE, Walstra FE, Mohammed MF, McLaughlin PD, Nicolaou S. Dual-energy CT in evaluation of the acute abdomen. Radiographics. 2019;39(1):264–86.
- Issa G, Mulligan M. Dual energy CT can aid in the emergent differentiation of acute traumatic and pathologic fractures of the pelvis and long bones. Emerg Radiol. 2020;27(3):285–92.
- Riveros Gilardi B, Munoz Lopez JI, Hernandez Villegas AC, Garay Mora JA, Rico Rodriguez OC, Chavez Appendini R, et al. Types of cerebral herniation and their imaging features. Radiographics. 2019;39(6):1598–610.
- Law M. Neurological complications. Cancer Imaging. 2009;9 Spec No A(Special issue A):S71–4.
- Guimaraes MD, Bitencourt AG, Marchiori E, Chojniak R, Gross JL, Kundra V. Imaging acute complications in cancer patients: what should be evaluated in the emergency setting? Cancer Imaging. 2014;14(1):18.
- Rogers LR. Cerebrovascular complications in patients with cancer. Semin Neurol. 2004;24(4):453–60.
- 11. Nemoto K, Aoshiba K, Itoh M, Semba S, Tsuji T, Adachi H, et al. Isolated pineal region metastasis from lung adenocarcinoma with obstructive hydrocephalus: a case report. J Med Case Rep. 2013;7:71.
- Clarke JL, Perez HR, Jacks LM, Panageas KS, Deangelis LM. Leptomeningeal metastases in the MRI era. Neurology. 2010;74(18):1449–54.
- Nayar G, Ejikeme T, Chongsathidkiet P, Elsamadicy AA, Blackwell KL, Clarke JM, et al. Leptomeningeal disease: current diagnostic and therapeutic strategies. Oncotarget. 2017;8(42):73312–28.
- Bruna J, Simo M, Velasco R. Leptomeningeal metastases. Curr Treat Options Neurol. 2012;14(4):402–15.
- Stam J. Thrombosis of the cerebral veins and sinuses. N Engl J Med. 2005;352(17):1791–8.
- Sasidharan PK. Cerebral vein thrombosis misdiagnosed and mismanaged. Thrombosis. 2012;2012:210676.
- Paner A, Jay WM, Nand S, Michaelis LC. Cerebral vein and dural venous sinus thrombosis: risk factors, prognosis and treatment—a modern approach. Neuro-Ophthalmology. 2009;33(5):237–47.
- Mut M, Schiff D, Shaffrey ME. Metastasis to nervous system: spinal epidural and intramedullary metastases. J Neuro-Oncol. 2005;75(1):43–56.
- Shah LM, Salzman KL. Imaging of spinal metastatic disease. Int J Surg Oncol. 2011;2011:769753.
- Olcott EW, Dillon WP. Plain film clues to the diagnosis of spinal epidural neoplasm and infection. Neuroradiology. 1993;35(4):288–92.
- 21. Cha S. Neuroimaging in neuro-oncology. Neurotherapeutics. 2009;6(3):465–77.
- 22. Dalmau J, Rosenfeld MR. Paraneoplastic syndromes of the CNS. Lancet Neurol. 2008;7(4):327–40.
- Williamson JP, Phillips MJ, Hillman DR, Eastwood PR. Managing obstruction of the central airways. Int Med J. 2010;40(6):399–410.
- Haas AR. Recent advances in the palliative management of respiratory symptoms in advanced-stage oncology patients. Am J Hosp Palliat Care. 2007;24(2):144–51.
- 25. Lewis MA, Hendrickson AW, Moynihan TJ. Oncologic emergencies: pathophysiology, presentation, diagnosis, and treatment. CA Cancer J Clin. 2011;61(5):287–314. Erratum in: CA Cancer J Clin. 2011 Nov-Dec;61(6):420. Dosage error in article text
- Theodore PR. Emergent management of malignancy-related acute airway obstruction. Emerg Med Clin North Am. 2009;27(2):231–41.
- 27. Shin JH, Song HY, Ko GY, Lim JO, Yoon HK, Sung KB. Esophagorespiratory fistula: long-term results of palliative treatment with covered expandable metallic stents in 61 patients. Radiology. 2004;232(1):252–9.
- 28. Balazs A, Galambos Z, Kupcsulik PK. Characteristics of esophagorespiratory fistulas resulting from esophageal cancers: a single-

center study on 243 cases in a 20-year period. World J Surg. 2009;33(5):994-1001.

- Katabathina VS, Restrepo CS, Martinez-Jimenez S, Riascos RF. Nonvascular, nontraumatic mediastinal emergencies in adults: a comprehensive review of imaging findings. Radiographics. 2011;31(4):1141–60.
- Harris JA, Bartelt D, Campion M, Gayler BW, Jones B, Hayes A, et al. The use of low-osmolar water-soluble contrast in videofluoroscopic swallowing exams. Dysphagia. 2013;28(4):520–7.
- Quint LE. Thoracic complications and emergencies in oncologic patients. Cancer Imaging. 2009;9 Spec No A(Special issue A):S75–82.
- Warner P, Uberoi R. Superior vena cava stenting in the 21st century. Postgrad Med J. 2013;89(1050):224–30.
- 33. Fagedet D, Thony F, Timsit JF, Rodiere M, Monnin-Bares V, Ferretti GR, et al. Endovascular treatment of malignant superior vena cava syndrome: results and predictive factors of clinical efficacy. Cardiovasc Intervent Radiol. 2013;36(1):140–9.
- Sheth S, Ebert MD, Fishman EK. Superior vena cava obstruction evaluation with MDCT. AJR Am J Roentgenol. 2010;194(4):W336–46.
- Barshes NR, Annambhotla S, El Sayed HF, Huynh TT, Kougias P, Dardik A, et al. Percutaneous stenting of superior vena cava syndrome: treatment outcome in patients with benign and malignant etiology. Vascular. 2007;15(5):314–21.
- 36. Ganeshan A, Hon LQ, Warakaulle DR, Morgan R, Uberoi R. Superior vena caval stenting for SVC obstruction: current status. Eur J Radiol 2009;71(2):343–349.
- Sakr L, Dutau H. Massive hemoptysis: an update on the role of bronchoscopy in diagnosis and management. Respiration. 2010;80(1):38–58.
- Chun JY, Morgan R, Belli AM. Radiological management of hemoptysis: a comprehensive review of diagnostic imaging and bronchial arterial embolization. Cardiovasc Intervent Radiol. 2010;33(2):240–50.
- Thirumaran M, Sundar R, Sutcliffe IM, Currie DC. Is investigation of patients with haemoptysis and normal chest radiograph justified? Thorax. 2009;64(10):854–6.
- Komissarova M, Chong S, Frey K, Sundaram B. Imaging of acute pulmonary embolism. Emergn Radiol. 2013;20(2):89–101.
- Klok FA, Mos IC, Kroft LJ, de Roos A, Huisman MV. Computed tomography pulmonary angiography as a single imaging test to rule out pulmonary embolism. Curr Opin Pulm Med. 2011;17(5):380–6.
- Weidman EK, Plodkowski AJ, Halpenny DF, Hayes SA, Perez-Johnston R, Zheng J, et al. Dual energy CT angiography for detection of pulmonary emboli: incremental benefit of iodine maps. Radiology. 2018;289(2):546–53.
- Retter AS. Pericardial disease in the oncology patient. Heart Dis. 2002;4(6):387–91.
- Bussani R, De-Giorgio F, Abbate A, Silvestri F. Cardiac metastases. J Clin Pathol. 2007;60(1):27–34.
- Peebles CR, Shambrook JS, Harden SP. Pericardial disease–anatomy and function. Br J Radiol. 2011;84(Spec No 3):S324–37.
- 46. Burazor I, Imazio M, Markel G, Adler Y. Malignant pericardial effusion. Cardiology. 2013;124(4):224–32.
- McCurdy MT, Shanholtz CB. Oncologic emergencies. Crit Care Med. 2012;40(7):2212–22.
- Wang ZJ, Reddy GP, Gotway MB, Yeh BM, Hetts SW, Higgins CB. CT and MR imaging of pericardial disease. Radiographics. 2003;23(Spec No):S167–80.
- Hayanga AJ, Bass-Wilkins K, Bulkley GB. Current management of small-bowel obstruction. Adv Surg. 2005;39:1–33.
- Tso DK, Avery LL, Lev MH, Kamalian S. Nivolumab-induced small bowel obstruction and perforation: a rare but life-threatening side effect of immunotherapy. Emerg Radiol. 2020;27(1):107–10.

- 51. Silva AC, Pimenta M, Guimaraes LS. Small bowel obstruction: what to look for. Radiographics. 2009;29(2):423–39.
- Santillan CS. Computed tomography of small bowel obstruction. Radiol Clin N Am. 2013;51(1):17–27.
- Heller MT, Khanna V. Cross-sectional imaging of acute abdominal conditions in the oncologic patient. Emerg Radiol. 2011;18(5):417–28.
- Ramanathan S, Ojili V, Vassa R, Nagar A. Large bowel obstruction in the emergency department: imaging spectrum of common and uncommon causes. J Clin Imaging Sci. 2017;7:15.
- Taourel P, Kessler N, Lesnik A, Pujol J, Morcos L, Bruel JM. Helical CT of large bowel obstruction. Abdom Imaging. 2003;28(2):267–75.
- 56. Kumar A, Muir MT, Cohn SM, Salhanick MA, Lankford DB, Katabathina VS. The etiology of pneumoperitoneum in the 21st century. J Trauma Acute Care Surg. 2012;73(3):542–8.
- Lagoutte N, Doussot A, Leung U, Facy O, Bastie JN, Rat P, et al. Perforation of bowel lymphoma: beware of atypical presentations. J Gastrointest Cancer. 2014;45(2):121–5.
- 58. Ko GY, Ha HK, Lee HJ, Jeong YK, Kim PN, Lee MG, et al. Usefulness of CT in patients with ischemic colitis proximal to colonic cancer. AJR Am J Roentgenol. 1997;168(4):951–6.
- Deepak P, Devi R. Ischemic colitis masquerading as colonic tumor: case report with review of literature. World J Gastroenterol. 2011;17(48):5324–6.
- Saba L, Mallarini G. Computed tomographic imaging findings of bowel ischemia. J Comput Assist Tomogr. 2008;32(3):329–40.
- Fulwadhva UP, Wortman JR, Sodickson AD. Use of dual-energy CT and iodine maps in evaluation of bowel disease. Radiographics. 2016;36(2):393–406.
- 62. Potts J, Al Samaraee A, El-Hakeem A. Small bowel intussusception in adults. Ann R Coll Surg Engl. 2014;96(1):11–4.
- Boudiaf M, Soyer P, Terem C, Pelage JP, Maissiat E, Rymer R. Ct evaluation of small bowel obstruction. Radiographics. 2001;21(3):613–24.
- Kim HC, Yang DM, Jin W, Park SJ. The various manifestations of ruptured hepatocellular carcinoma: CT imaging findings. Abdom Imaging. 2008;33(6):633–42.
- Lucey BC, Varghese JC, Soto JA. Spontaneous hemoperitoneum: causes and significance. Curr Prob Diagn Radiol. 2005;34(5):182–95.
- 66. Sethi V, Philips S, Fraser-Hill M. Lines and circles: pictorial review of cross-sectional imaging of active bleeding and pseudoaneurysm in the abdomen and pelvis. Can Assoc Radiol J. 2013;64(1):36–45.
- Allen DJ, Longhorn SE, Philp T, Smith RD, Choong S. Percutaneous urinary drainage and ureteric stenting in malignant disease. Clin Oncol (R Coll Radiol). 2010;22(9):733–9.

- Silverman SG, Leyendecker JR, Amis ES Jr. What is the current role of CT urography and MR urography in the evaluation of the urinary tract? Radiology. 2009;250(2):309–23.
- Ishioka J, Kageyama Y, Inoue M, Higashi Y, Kihara K. Prognostic model for predicting survival after palliative urinary diversion for ureteral obstruction: analysis of 140 cases. J Urol. 2008;180(2):618– 21. discussion 21
- Kogut MJ, Bastawrous S, Padia S, Bhargava P. Hepatobiliary oncologic emergencies: imaging appearances and therapeutic options. Curr Prob Diagn Radiol. 2013;42(3):113–26.
- Huggett MT, Ghaneh P, Pereira SP. Drainage and bypass procedures for palliation of malignant diseases of the upper gastrointestinal tract. Clin Oncol (R Coll Radiol). 2010;22(9):755–63.
- 72. Shrikhande SV, Barreto SG, Goel M, Arya S. Multimodality imaging of pancreatic ductal adenocarcinoma: a review of the literature. HPB (Oxford). 2012;14(10):658–68.
- 73. Romagnuolo J, Bardou M, Rahme E, Joseph L, Reinhold C, Barkun AN. Magnetic resonance cholangiopancreatography: a meta-analysis of test performance in suspected biliary disease. Ann Intern Med. 2003;139(7):547–57.
- van Delden OM, Lameris JS. Percutaneous drainage and stenting for palliation of malignant bile duct obstruction. Eur Radiol. 2008;18(3):448–56.
- 75. Coleman RE. Clinical features of metastatic bone disease and risk of skeletal morbidity. Clin Cancer Res. 2006;12(20 Pt 2):6243s–9s.
- 76. Trinkaus M, Simmons C, Myers J, Dranatisaris G, Clemons M. Skeletal-related events (SREs) in breast cancer patients with bone metastases treated in the nontrial setting. Support Care Cancer. 2010;18(2):197–203.
- Hwang S, Panicek DM. Magnetic resonance imaging of bone marrow in oncology, part 1. Skelet Radiol. 2007;36(10):913–20.
- Trumm CG, Jakobs TF, Zech CJ, Helmberger TK, Reiser MF, Hoffmann RT. CT fluoroscopy-guided percutaneous vertebroplasty for the treatment of osteolytic breast cancer metastases: results in 62 sessions with 86 vertebrae treated. J Vasc Intervent Radiol. 2008;19(11):1596–606.
- Mirels H. Metastatic disease in long bones. A proposed scoring system for diagnosing impending pathologic fractures. Clin Orthop Relat Res. 1989;249:256–64.
- Deschamps F, Farouil G, Hakime A, Barah A, Guiu B, Teriitehau C, et al. Cementoplasty of metastases of the proximal femur: is it a safe palliative option? J Vasc Intervent Radiol. 2012;23(10):1311–6.
- Jawad MU, Scully SP. In brief: classifications in brief: Mirels' classification: metastatic disease in long bones and impending pathologic fracture. Clin Orthop Relat Res. 2010;468(10):2825–7.

Ultrasound



47

Sherwin Soltani, Konrad Karasek, Lamees M. Alshaikh, Ideen Zeinali, and Jennifer Carnell

Introduction

The use of bedside ultrasound by trained physicians has advanced the practice of medicine. Over the past two decades, clinician-performed ultrasound has seen widespread adoption as a core skill in emergency medicine and critical care medicine [1, 2]. With appropriate training and a symptom-based approach, emergency ultrasound is fieldtested in the rapid diagnosis and management of undifferentiated critically ill patients [3–5]. The numerous emergencies associated with cancer and its treatment, including septic shock, ascites, pleural and pericardial effusions, pulmonary embolism, deep venous thrombosis, and bowel obstruction, among others, can all be diagnosed with bedside ultrasound, helping the clinician focus their initial differential diagnosis, thus decreasing time to diagnosis and stabilization.

Bedside ultrasound, also known as point-of-care ultrasound (POCUS), emergency ultrasound, or clinical ultrasound, is an extremely broad topic, and the interested reader is encouraged to learn more about this topic through the many available resources in textbooks and free-and-open access medical education.

Basic Ultrasound Functionality

An ultrasound machine is always paired with a transducer that is applied to the patient's skin. The transducer contains piezoelectric crystals or a specialized transducer-on-chip that vibrates when electricity is applied to it, generating ultrasound waves typically in the range of 2–15 MHz that are conducted through the body through a coupling medium such as gel [6, 7]. Soft tissue, blood, simple fluid, fat, bone, and air all conduct these ultrasound waves with slightly different velocities and impedance, producing reflections that travel back to the transducer. These reflections are processed by the machine based on how long they take to return, where on the ultrasound array they are heard, how strong they are and their frequency, to produce a 2D representation called a B-mode image (Fig. 47.1a) [7, 8].

Bright or hyperechoic areas of B-mode images represent stronger reflectors of ultrasound waves such as bone. Dark or hypoechoic areas of the image are the result of weaker reflectors. Areas where there are no ultrasound wave reflections, as seen with fluid, are termed anechoic and appear black on the ultrasound screen. The resolution of an ultrasound image is a function of the machine's sensitivity and the frequency used to generate the image. Higher frequency ultrasound waves produce higher resolution images at the cost of decreased depth of penetration, while lower frequency ultrasound can image deeper structures but sacrifices resolution.

Modern ultrasound machines incorporate other functionality. M-mode images such as Fig. 47.1b are formed by taking a single "slice" of a B-mode image and graphing motion within that slice over time and are useful for imaging salient parts of rapidly moving structures, such as the heart [8]. Doppler imaging, so named for the physics concept behind the technology, measures the difference in returned ultrasound frequency to determine the velocity of fluid within the image. Color Doppler imaging (Fig. 47.1c) employs color over a grayscale image to indicate direction of flow within a structure, while spectral Doppler measures precise flow velocities and direction of flow within a structure.

Ultrasound Transducer Selection

Ultrasound transducers vary in their frequency range and the method they use to steer ultrasound waves. Usually, these transducers are marked on one side with a transducer marker, which may be a small notch or coloration. This corresponds to a dot or

S. Soltani · K. Karasek · L. M. Alshaikh · I. Zeinali · J. Carnell (⊠) Department of Emergency Medicine, Ben Taub General Hospital, Baylor College of Medicine, Houston, TX, USA



Fig. 47.1 Ultrasound imaging modes. (a) B-mode, also called 2D mode on some machines, produces 2D image of echo-producing interfaces of underlying objects. (b) M-mode, "motion mode," demonstrates movement of structures over time. All cardiac structures through which the solid white line passes in the top B-mode image are graphed over

another marker on the ultrasound machine display, allowing the operator to anatomically orient their image on the screen.

Linear array transducers (Fig. 47.2a), as their name implies, use crystals arranged in a line, firing sequentially to produce an image which represents a slice of the overall area

time in the bottom M-mode image. With this mode, precise measurements can also be made. (c) Color Doppler mode shows the direction of blood flow in a color map superimposed on this 2D image of the bifurcation of the common carotid artery

being imaged [6]. These high-frequency transducers produce images with excellent resolution but have limited depth of penetration. Consequently, they are primarily used for imaging superficial structures or masses, evaluating veins for thrombosis or for real-time procedural guidance.



Fig. 47.2 Commonly employed ultrasound transducers. (a) Linear array transducer (high frequency, 4–12 MHz), (b) curvilinear array transducer (low frequency, 1–5 MHz), and (c) phased-array transducer (low frequency, 1.5–4 MHz)

Curvilinear arrays are convex-shaped crystal arrays (Fig. 47.2b). Their larger footprint creates a wider field of view, allowing them to evaluate larger organs more easily. These transducers typically have lower frequency output and are suitable for imaging the abdomen, thorax, or deeper extremity structures.

Phased-array transducers (Fig. 47.2c) have a smaller footprint. Instead of using long arrays of crystals, a smaller number of crystals are instead pulsed in specific timing sequences that allow the beam to be steered electronically. These transducers produce pie-shaped images that are similar to curvilinear arrays. A significant advantage of their reduced footprint is that it allows imaging of areas of the body where acoustic windows are limited due to underlying air or bone, most notably in cardiac imaging. Phased-array transducers are also capable of general abdominal and thoracic imaging.

Endocavitary transducers commonly produce midrange ultrasound frequencies and are arranged as a highly convex curvilinear array. In combination with appropriate institutional protocols for high-level disinfection and sterile transducer covers, these transducers can be used for obtaining intraoral, transvaginal, or transrectal images that can aid in the visualization of certain organs or pathologies.

Adjusting the Ultrasound Image

On most machines, the operator has direct control over the depth, gain, and the imaging preset or resolution. Machines vary in their capabilities for more advanced functionality, such as Doppler imaging, M-mode imaging, zooming, frequency adjustment, or harmonic imaging. Depth has a dedicated control and is indicated to the side of the image by hash marks, and often the maximal depth is displayed numerically next to the image. Gain controls how bright or dark the image appears on the screen and can be adjusted for the entire image at once or can be adjusted at certain depths, depending on the machine's capabilities, a function called time gain compensation (TGC) [7].

Most ultrasound machines have imaging presets, for example, an abdominal, cardiac, or vascular mode. Presets may adjust frequency settings, output power, image postprocessing, or the frame rate to help optimize the images displayed.

Safety of Bedside Ultrasound

The amount of ultrasound energy that is imparted during an exam should be minimized in accordance with the ALARA principle. Although ultrasound does not use ionizing radiation, ultrasound waves can induce effects such as cavitation, causing local tissue damage or thermal heating. These effects are displayed by the machine in terms of mechanical index (MI) and thermal index (TI), respectively, usually as unitless numbers. In general, the lowest numbers achievable with the shortest duration of imaging minimize the potential for adverse effects [8]. Tissue heating and mechanical disruption are of most concern for obstetric imaging and for imaging of the eye.

Other safety considerations include infection control, the need for sterile probe cover use during invasive procedures, and guidelines for disinfection of transducers, all of which are usually established by local institutional guidelines and protocols.

Emergency Cardiac Ultrasound

Basic Approach

There are four essential views for transthoracic cardiac ultrasound: the parasternal long axis, the parasternal short axis, the apical four chamber, and the subxiphoid. A small footprint phased-array transducer is used with the cardiac preset and a depth of 16–20 cm. A curvilinear array transducer may be used if the phased-array transducer is not available.

For the parasternal views, the transducer marker is aimed toward the right shoulder or the right hip for the long and short axis, respectively. In the apical four-chamber and the subxiphoid views, the transducer marker is aimed toward patient's right (Fig. 47.3). Cardiac visualization may be improved by placing the patient in the left lateral decubitus position and instructing the patient to hold their breath after exhaling. The subxiphoid view may be improved by having the patient flex their hips and knees and inhale slightly.

Pericardial Effusion and Cardiac Tamponade

Ultrasonography is the imaging modality of choice for detecting pericardial effusions and tamponade, common abnormalities found in oncologic patients, which may be a consequence of the underlying cancer diagnosis [9] or secondary to chemotherapy [10, 11]. In the normal heart, the pericardium appears as a hyperechoic outline of the more hypoechoic myocardium. A pericardial effusion appears between these two layers [12, 13]. An effusion with simple fluid typically appears as an anechoic stripe (Fig. 47.4; see Fig. 47.4b). However, complex malignant, exudative, or hemorrhagic effusions may have hyperechoic portions or contain septations.

The pericardial fat pad may mimic a small pericardial effusion. Frequently, this more solid component appears to contain internal echogenicity and moves with the epicardium to which it is attached in time with the cardiac cycle. Multiple cardiac views may be necessary to differentiate pericardial fat from an effusion [12].

Pericardial effusion is best detected in the subxiphoid and parasternal views by interrogating the most dependent potions of the pericardial sac where fluid is expected to collect first. The most dependent area in the subxiphoid view lies anterior and adjacent to the right ventricle and, in the parasternal views, lies along the posterior aspect of the pericardial sac, just anterior to the descending thoracic aorta (see Fig. 47.4b) [13]. In the apical view, fluid may appear anywhere circumferentially. While fluid in the pericardial sac preferentially collects in the most dependent area, in complex malignant, exudative, or hemorrhagic effusions, fluid may defy this rule and collect in nondependent areas.

Cardiac tamponade is a crucial diagnosis as it is a potentially reversible cause of shock. POCUS allows the realtime visualization of the effects of increased intrapericardial pressure on the heart. The right side of the heart typically



Fig. 47.3 Transducer placement for cardiac views. (a) Parasternal views, transducer normally placed in the fifth intercostal space to the left of the sternum. For the parasternal long-axis view, the transducer marker is oriented to the right shoulder, and for the short-axis view, it is rotated 90° to the right hip. (b) Apical view, with the transducer marker

has the lowest intracardiac pressure and thus is affected first. The right ventricle (RV) and right atrium (RA) are best visualized in the apical four-chamber or subxiphoid views [12, 14]. Inversion of the RA occurring during systole (Fig. 47.5a), when atrial filling should occur, is the most sensitive, though least specific, sign of cardiac tamponade. As pericardial pressure increases, RV free wall inversion (RVI) can be noted during diastole (Fig. 47.5b) [12, 14]. M-mode imaging is of utility, as most patients with tamponade are tachycardic, making visual detection of signs of tamponade challenging. By positioning the cursor through

to the patient's right, typically placed at the midclavicular line at the level of the nipple line or at the mammary fold in female patients. (c) Subxiphoid view, with the transducer marker to the patient's right, transducer placed under the xiphoid process, aimed slightly to the left hemithorax

the mitral valve (MV) leaflets and the RV free wall (Fig. 47.5c), diastole can be identified as the portion of the cardiac cycle when the MV leaflets are open and concurrent RVI may be detected (Fig. 47.5d).

In tamponade, elevated central venous pressure results in a dilated IVC that has minimal respirophasic variation.

In the rapidly deteriorating patient, derangement of vital signs may preclude in-depth cardiac examination, and a finding of pericardial effusion with clinical symptoms of tamponade should prompt emergent intervention by the clinician.



Fig. 47.4 Parasternal and subxiphoid views of the heart. (**a**) Normal parasternal long-axis view. (**b**) Parasternal long axis with circumferential pericardial effusion (*arrows*). (**c**) Normal subxiphoid view. (**d**)

Subxiphoid view with effusion (*arrows*). LA, left atrium; LV, left ventricle; LVOT, left ventricular outflow tract; RA, right atrium; RV, right ventricle; DTA, descending thoracic aorta

Left Ventricular Function

Left ventricular systolic function can be rapidly estimated by a bedside echocardiogram. In the normal left ventricle (LV), there is simultaneous inward movement and thickening of the ventricular septal and free walls during systole (Fig. 47.6). Dilation, myocardial thinning, and poor contractility may all be observed in heart failure. Various methods of measuring the LV cavity size are known; however, visual estimation of the LV systolic function by emergency physicians skilled in bedside ultrasound has been shown to be comparable in performance to advanced measurements and estimation by cardiologists [15].

In patients without LV systolic impairment or MV disease, the MV opens briskly during early diastole, and the anterior leaflet, easily seen on the parasternal long-axis and apical views, moves toward the septal wall. The point at which the anterior MV leaflet is the closest to the septum is called the E-point, and the distance between the anterior MV leaflet and the septal wall is the E-point septal separation (EPSS). An EPSS of greater than 7 mm is highly sensitive for decreased LV systolic function, and in general the greater the EPSS, the worse the LV ejection fraction (LVEF) [16, 17]. The use of EPSS as an indicator of LV systolic function has not been validated in patients with valvular disease of the mitral or aortic valves, where impaired valve mobility may lower EPSS independent of LV function.

Right Ventricular Strain

Pulmonary embolus (PE) is one of the leading complications of malignancy and remains high on the differential of the cancer patient presenting with acute chest pain, shortness of breath, or hypotension. Massive or sub-massive PEs are a cause of obstructive shock and rapid decompensation [18].

With POCUS, the RV is best evaluated in the apical fourchamber view. RV dilation is variably defined; however, a



Fig. 47.5 Echocardiographic signs of tamponade. (a) Right atrial inversion (RAI) seen during systole (MV closed) in the apical fourchamber view, a more sensitive but less specific sign. (b) Diastolic right ventricular inversion (RVI), a more specific sign of cardiac tamponade.

ratio of RV/LV > 0.9 is considered abnormal [19, 20]. In the normal parasternal short-axis view, the LV has a uniform circular shape, with the septum pushed outward into the RV due to the higher chamber pressure within the LV relative to the RV.

In the setting of RV strain, there is dilation of the right ventricle. The increasing pressure within the RV can manifest as septal flattening, where the normal curvature of the septum into the RV is lost. This finding is known as the "D-sign" on the parasternal short-axis view and will initially be seen in diastole but progress to be present throughout the entire cardiac cycle (Fig. 47.7; see 47.7a). On the apical fourchamber view, RV free wall hypokinesis with relatively preserved apical motion, a finding known as McConnell's sign, may be seen [21]. Anterior movement of the tricuspid annulus during RV systole, a measurement known as tricuspid

(c) M-mode through the anterior mitral valve leaflet and RV outflow tract in the parasternal long-axis view, with (d) showing the resulting M-mode tracing. When the MV is open (*) in diastole, there is inward movement of the RV free wall (*arrowhead*)

annular plane of systolic excursion (TAPSE), is a good surrogate for RV systolic function in the typical adult heart (Fig. 47.8) [22]. This can be measured using an M-mode cursor through the junction of the tricuspid annulus and the RV free wall. In general, an abnormal TAPSE is <17 mm [22], and in confirmed PE, TAPSE <15 mm confers increased risk of mortality or need for thrombolysis [19, 20]. Interrogation of the IVC may reveal dilation without respirophasic variation, suggestive of increased central venous pressure.

Measurement of the Inferior Vena Cava

The inferior vena cava (IVC) can be seen from a subcostal approach as it courses posterior to the liver before passing above the diaphragm. From a subxiphoid view of the heart,


Fig. 47.6 Left ventricular ejection fraction evaluated in the parasternal short axis. (a, b) End systole and end diastole in a normal heart. (c, d) End systole and end diastole, with markedly reduced wall contraction

resulting in little to no change in the left ventricular size. (*) Small left pleural effusion



Fig. 47.7 Elevated RV pressure resulting in echocardiographic changes. (a) The "D-sign": RV strain produces septal flattening (*arrow*) on the parasternal short-axis view, which results in a loss of the typical cylindrical shape of the LV. (b) RV and RA dilation in the setting of

elevated right heart pressures. In end diastole, the RV cavity size measured near the base is larger than the LV size, and the septum is pushed toward the left heart (*arrowheads*)

Fig. 47.8 Tricuspid annular plane systolic excursion (TAPSE) measurement. The M-mode cursor is positioned through the junction of the tricuspid annulus and RV free wall in the apical fourchamber view. The resulting M-mode tracing is measured peak to trough. (**a**) Normal TAPSE (32 mm). (**b**) Decreased TAPSE in a patient with acute right heart strain (17 mm)



the transducer should be fanned posteriorly until the IVC is identified near its junction with the right atrium. The transducer can then be turned to a long-axis view with the transducer marker cephalad. Direct visualization of the IVC in the supine, spontaneously breathing patient, assuming there is no IVC obstruction or altered anatomy, correlates to a limited extent with the central venous pressure (CVP) or right atrial pressure (RAP). Respirophasic variation in the IVC size is due to changes in intrathoracic pressure caused by the respiratory cycle.

(*IVCmax*) (CI). Significant heterogeneity in research methods used by studies to correlate CVP to IVCmax or CI exists, though guidelines suggest that IVCmax >2.1 cm with CI < 0.50 (with sniffing maneuver) suggests an approximate RAP of 15 mmHg [22, 23]. More qualitatively, a dilated IVC with little respiratory variation may be seen in obstructive pathologies such as cardiac tamponade, pulmonary hypertension/ PE, RV failure, or tension pneumothorax (Fig. 47.9).

Fig. 47.9 Measurement of the inferior vena cava in the subcostal view.

The liver overlies the IVC, and typically the hepatic vein can be seen

emptying into the IVC (*). (a) Dilated IVC measuring greater than

2 cm, with low respiratory phasic variation (almost no collapse with

Thoracic and Pulmonary Oncologic Emergencies

POCUS allows medical providers to perform real-time, bedside evaluation of undifferentiated oncologic patients with respiratory symptoms in the emergency department. It is rapid, reproducible, and often more sensitive and specific than a radiograph.

Lung ultrasound can be performed with any of the commonly accessible transducers - curvilinear, phased-array, or linear – and selection will depend on the specific application as well as provider preference. For most applications, a lowfrequency curvilinear or phased-array transducer is best suited for detecting oncologic emergencies. Conventionally, the transducer marker is either oriented to the patient's right or toward their head. The exam mode should utilize the "pulmonary" or "lung" preset, and the depth of the image should be set to at least 18 cm. If a "lung" preset is not available, an "abdominal" preset is appropriate.

Zones of the Lung

Each hemithorax is divided into four distinct areas, for a combined eight zones on the anterolateral chest (Fig. 47.10). The lung assessment should be performed in a systematic and comprehensive manner. With the patient

inspiration) and thus a low collapsibility index (CI). The CVP in this patient is likely elevated. (b) Small IVC with CI of greater than 0.50, suggesting normal CVP

S. Soltani et al.

Fig. 47.10 Each hemithorax is divided into four distinct areas, for a combined eight zones on the anterolateral chest. Each field should be interrogated as part of the lung ultrasound exam. The PLAPS point (posterolateral alveolar and/or pleural syndrome) is the area where most pneumonias and consolidations will be visualized

supine or semi-recumbent, each hemithorax should be imaged in a sagittal orientation, with the transducer marker toward the patient's head. Landmarks include the pleural line (parietal and visceral pleura) and the lung parenchyma flanked by the ribs and their corresponding posterior acoustic shadowing. The angle of insonation should be manipulated to optimize the clarity of the pleural line (Fig. 47.11). Experts also recommend imaging the posterolateral alveolar and/or pleural syndrome (PLAPS) point, where most pneumonias and consolidations will be visualized [24, 25].





а



Fig. 47.11 The pleural line (*arrows*) appears as a bright hyperechoic line separating the superficial soft tissue from the lung parenchyma. The pleural line is flanked by hyperechoic ribs (+) with posterior acoustic shadowing (*). Utilizing an angle of insonation perpendicular to the pleural line will improve visualization of relevant findings. Lung sliding appears as a shimmering artifact at the pleural line and corresponds with the respiratory cycle

Lung Artifacts (A-Lines and B-Lines)

In normally aerated lung, there are linear, echogenic, horizontally oriented reverberations generated from the pleural line. These reverberation artifacts are called A-lines [26–28]. While A-lines are seen in normal lung, they are also seen in the setting of lung pathology such as chronic obstructive pulmonary disease (COPD), asthma, and pneumothorax. In contrast, B-lines are hyperechoic, discrete, vertical beams of artifact that generally obliterate A-lines, extend to the bottom of the screen, and move in synchrony with the patient's pleural line [26, 27]. B-lines represent increased density in the lung parenchyma and, depending on the clinical context, can be diagnostic of multiple pathologies relevant to malignancy, including tumor, pneumonia, pneumonitis, lung infarction, and fibrosis, among others (Fig. 47.12).

Diffuse Interstitial Syndrome

Looking in each of the eight scanning zones, a zone is positive if it has three or more B-lines at any point in a respiratory cycle. The presence of two or more positive zones, on both hemithoraces, is consistent with diffuse interstitial syndrome (DIS) [26, 27].

In general, DIS is most often due to cardiogenic pulmonary edema in the setting of heart failure (HF). One study showed that the combination of LVEF <45%, a distended non-collapsible IVC, and presence of DIS is 100% specific for the diagnosis of acute decompensated HF [29]. However, it is important to note that this is not always the case as DIS itself is nonspecific and may represent other causes of dyspnea. In cases of DIS where the triad above are not found, one should consider other causes such as HF with preserved LVEF, acute respiratory distress syndrome (ARDS), pulmonary fibrosis, diffuse parenchymal lung disease, pneumonitis secondary to chemotherapeutic agents, or multifocal pneumonia secondary to immunocompromised state. Additionally, the presence of unilateral or focal B-lines may also indicate any of the following: pneumonia, atelectasis, mass, or infarctions (mostly due to pulmonary embolism). As with all POCUS, integration of the entire clinical picture by the operator improves the accuracy of the diagnosis.

Consolidation

Patients with malignancy are often on chemotherapeutic agents and medications that cause immunocompromised state, putting them at higher risk for infection. Oncologic patients presenting with dyspnea are at high risk for pneumonia, which account for approximately 10% of their hospital admissions, with rates as high as 30% in patients with hematologic malignancies [30]. Additionally, prior surgeries, radiation, infections, or compressive masses can cause recurrent post-obstructive pneumonia due to the patient's inability to clear static secretions in bronchi and alveoli distal to the obstruction. This leads to atelectasis and subsequent microbial colonization, infection, and consolidation [30].

Pneumonia occurs on a spectrum, depending on how much of the aerated lung is replaced with fluid, pus, or increased density. This can progress between various degrees of focal B-lines, atelectasis, subpleural consolidation, and, ultimately, hepatization (lung taking on a tissue-like appearance of solid organs, Fig. 47.13). The presence of subpleural consolidation is demonstrated as an irregular pleural line with hypodensities adjacent and just deep to it and may indicate an infectious process (this finding is known as the shred sign, Fig. 47.14). The presence of static and/or dynamic air bronchograms can help distinguish between atelectasis and pneumonia. Static air bronchograms are caused by the air that remains within the bronchi in a consolidative process and appear as nonmobile hyperechoic lines extending out to the periphery of the lung signifying complete bronchial obstruction. Static air bronchograms may be seen in resorptive atelectasis and pneumonia and, therefore, cannot be used to differentiate these pathologies (see Fig. 47.13). Dynamic air bronchograms are caused by hyperechoic air moving toward the periphery of the lung with inspiration [31]. Dynamic air bronchograms have a 94% specificity and a 97% positive predictive value for diagnosing pneumonia and distinguishing it from resorptive atelectasis [31].

Fig. 47.12 A-lines are linear, equidistant, echogenic, horizontally oriented reverberation artifacts generated from the pleural line (*horizontal arrows*). B-lines result from increased density in the lung parenchyma and are hyperechoic, discrete, vertical beams that extend from and move with the pleural line, reaching a depth of 18 cm. B line artifacts (*) generally obliterate A-lines





Fig. 47.13 Hepatization (#) represents highly consolidated lung parenchyma. It has similar echogenicity to the liver (*). The liver and consolidated lung are separated by the diaphragm (line). Within the consolidation, one can see static and/or dynamic air bronchograms (*arrows*)



Fig. 47.14 Subpleural consolidation (*outlined*) is demonstrated as an irregular pleural line with subpleural hypodensities. It may indicate an infectious process. A normal pleural line (*arrows*) is also demonstrated here

Pleural Effusions

Pleural and parapneumonic effusions are common causes of dyspnea and chest pain in oncologic patients and have a significant effect on quality of life, often requiring frequent visits to the emergency department and procedures (i.e., thoracentesis) to help manage symptoms. While chest radiographs are traditionally used to diagnose effusion, POCUS is much more sensitive and specific, reliably identifying effusions as small as 20 mL, and is 100% sensitive for effusions >100 mL [32]. Pleural effusions are diagnosed by noting the absence of the mirror artifact (the echogenic reflection of the liver or spleen above the diaphragm) and by visualizing the spine sign (thoracic vertebral bodies extending cephalad past the diaphragm), which is approximately 93% sensitive and specific for detecting non-trace pleural effusions (Fig. 47.15) [33]. Effusions can be simple and anechoic, or complex and heterogeneous, sometimes mimicking consolidations or malignant effusions. POCUS can help distinguish whether the effusions are unilateral, bilateral, loculated, diffuse, simple, or complex. One study showed that ultrasonographic characteristics could help distinguish between transudative and exudative effusions. In this paper, they showed that transudates were *always* anechoic, whereas effusions with complex septated, complex non-septated, or homogeneously echogenic patterns were always exudative [34]. Additionally, POCUS can help identify relevant anatomy to help guide thoracentesis for recurrent malignant effusions and decreases the rate of pneumothorax from approximately 30% (landmark-based approach) to less than 3% [35].

Pleural Disease

The pleural line is best evaluated using the linear transducer, allowing for better resolution of the superficial structures.



Fig. 47.15 This image shows an anechoic pleural effusion (*). Pleural effusions most commonly appear as anechoic collections above the diaphragm (line). Additional findings include absence of the mirror artifact (the echogenic reflection of the liver or spleen above the diaphragm) and the spine sign (visualization of thoracic vertebral bodies extending cephalad past the diaphragm) (*arrows*)

Normal pleura is smooth, hyperechoic, and thin (no more than 3 mm) [32]. The parietal and visceral pleura generally cannot be distinguished unless there is pathology present. Thickening can be caused by multiple etiologies, including infection and malignancy. Parietal pleura thickening >10 mm is predictive of malignancy [32]. Diffuse pleural irregularities are likely due to generalized processes like pulmonary fibrosis or pneumonitis, which may indicate an adverse reaction to chemotherapeutic agents in oncologic patients.

Pneumothorax

In patients with malignancy, iatrogenic causes (e.g., thoracentesis, biopsy, surgery) and the malignancy itself can cause pneumothorax. Approximately 10% of these patients will go on to have a malignancy-associated secondary spontaneous pneumothorax which is associated with increased morbidity and mortality [36].

POCUS is far superior to physical exam and chest radiograph for identifying pneumothoraces [37]. Pneumothoraces can be identified fairly quickly and easily with POCUS by placing a high-frequency linear transducer on the most superior portion of the anterior chest, with the transducer marker toward the head. Both sides of the chest should be evaluated. Start by looking for evidence of lung sliding between the parietal and visceral pleura. The appearance of lung sliding on ultrasound is best described as a "shimmering" of the pleural line or "ants crawling" along the pleural line and is produced by the apposed movement of the parietal and visceral pleura. Additionally, the presence of B-lines or comet tails at the pleural line is highly sensitive for ruling out pneumothorax [26, 27, 38]. The presence of lung point is the most specific (near 100%) for diagnosing pneumothorax [37].

Abdominal Oncologic Emergencies

Approximately 40% of cancer patients present to the emergency department with gastrointestinal symptoms [39]. POCUS can help evaluate for some of the more common pathologies.

Malignant Bowel Obstruction

Malignant bowel obstruction (MBO) usually results from mechanical obstruction caused by tumor burden, radiationinduced fibrosis, adhesive disease, or dysfunctional bowel motility. The incidence of MBO is approximately 3–15% of all patients with advanced cancer and occurs at all levels of the GI tract – gastric outlet, small bowel (most frequent), and large bowel [39].

In the supine position, a low-frequency curvilinear or phased-array transducer is used to scan the abdomen in both the transverse and sagittal planes, moving up and down the abdomen in the systematic lawn-mower pattern from one lower quadrant to another. The presence of a small bowel obstruction (SBO) is detected by identifying dilated (>2.5 cm, most sensitive), noncompressible, loops of bowel with the presence of back-and-forth movement of intraluminal contents (most specific, Fig. 47.16). Additional evidence of a bowel obstruction can be indicated by an increase in bowel wall thickness (>3 mm) or the presence of localized intraperitoneal free fluid tracking between the loops of bowel (tanga sign). POCUS is between 91-98% sensitive and 93-98% specific, for the diagnosis of SBO when performed by providers with minimal training [40, 41]. Large bowel obstruction has similar features but is generally dilated to >5.0 cm.

Ascites and Perforation

One of the most common complications that providers will encounter will be the presence of malignant ascites or free fluid in the abdomen – usually secondary to hepatobiliary, gynecologic, or metastatic malignancies. Specific views include the space between the liver and kidney (hepatorenal space or Morison's pouch), the area around the spleen (perisplenic space), and the area around and behind the bladder (retro-vesicular/rectovaginal space or pouch of Douglas) – these are components of the well-known FAST exam (focused assessment with sonography in trauma) [28]. A dark or anechoic area in any of these three potential spaces

S. Soltani et al.

Fig. 47.16 Examples (a) and (**b**) of a small bowel obstruction (SBO). An SBO is detected by identifying dilated (>2.5 cm, most sensitive, double-headed dashed line). noncompressible loops of bowel with the presence of intraluminal contents without visible forward movement with peristalsis ("to-and-fro peristalsis," most specific. solid double-headed arrow). Additionally, increased bowel wall thickening, >3 mm (small arrows) (a), and localized free intraperitoneal fluid (#) or fluid tracking between the distended loops ("tanga sign") may also be seen (*) as in (b)



indicates free intraperitoneal fluid, which may represent ascites, blood, or bowel contents, depending on the clinical picture. Similar to thoracic pleural effusions, ascites fluid may be simple, complex, infectious, loculated, or debris filled. Again, a low-frequency curvilinear or phased-array transducer is generally used to interrogate for free fluid within the abdomen. With the patient supine, and with the transducer marker toward the head, the transducer is placed on the right and left flanks, approximately at the ninth/tenth intercostal space and angled in a slightly oblique angle to insonate between the rib spaces. This provides a view of the right and left hemi-abdominal potential spaces [28]. The final portion is the pelvic view, which is generally performed in both the transverse and sagittal planes, interrogating the retro-vesicular space in men and the rectovaginal space/pouch of Douglas in women [28]. The FAST exam has been reported to detect intraperitoneal fluid collections as small as 100 mL, with a median range of 250–620 mL commonly cited. The overall sensitivity and specificity of the FAST exam have been reported to be approximately 79% and 99%, respectively [42]. Similar to thoracentesis, ultrasound-guided paracentesis has been associated with a 68% decrease in bleeding complications when obtaining samples to better characterize the intra-abdominal fluid [35].

Immunocompromised oncologic patients are at increased risk for bowel perforation due to local infiltration, vascular occlusion, or increased pressures from obstructive processes causing ischemia and necrosis of the intestinal wall [39]. Additionally, opportunistic infections such as cytomegalovirus, chemo-/radiotherapy, and steroids have been linked with spontaneous perforation [39]. In perforation, complex free fluid that is heterogeneous and echogenic may be visualized outside the luminal walls of the intestinal tract.

Colitis

Despite the benefits of new cancer treatments, many novel therapies including immune checkpoint inhibitors can lead to autoimmunity and immune-related adverse events resulting in significant morbidity [39]. Autoimmune processes, including graft-versus-host disease, and opportunistic gastrointestinal infections (e.g., cytomegalovirus or Clostridium difficile), can all lead to colitis. Neutropenic enterocolitis (NEC) is a common presentation of abdominal pain in oncologic patients, representing a spectrum of disease, which includes typhlitis, caused by inflammation of the gastrointestinal tract. Chemotherapy and neutropenia weaken the mucosal barrier allowing bacterial translocation across the intestinal wall, mucosal damage, intestinal hemorrhage, transmural necrosis, and local infiltration. NEC can mimic an acute surgical abdomen and is considered a diagnosis of exclusion. Ultrasound may help detect NEC by identifying bowel wall thickening (BWT). Clinically, the presence of BWT is associated with increased duration of symptoms and mortality [43]. Generally accepted criteria for diagnosis include BWT >4 mm (transverse scan) in any segment of the bowel for >3 cm in length (longitudinal scan) [44]. BWT >10 mm is associated with more severe disease and increased mortality [43]. The three mural layers of the gastrointestinal tract are often visualized as they become more edematous and generally have a distinct hyper-hypo-hyperechoic pattern [43]. Ultrasound imaging has also been found to be useful in monitoring the clinical response of patients with NEC by demonstrating measurable reductions in BWT. However, ultrasound does have some disadvantages compared to CT, including limited resolution in obese patients, operator dependency/training, lower sensitivity, and inability to completely evaluate or exclude other potential surgical causes of abdominal pain [44].

Genitourinary Oncologic Emergencies

Obstructive Urinary Pathologies

Patients with prostatic, vesicular, gynecologic, rectal, or intra-abdominal malignancies, metastases, or retroperitoneal

lymphadenopathy/fibrosis are at risk for localized compression or obstruction of genitourinary structures which may, in turn, cause urinary retention, hydronephrosis, hematuria, anuria, and acute kidney injury or failure. The use of POCUS for the early identification of hydronephrosis or genitourinary lesions may guide management (e.g., radiation or decompression via nephrostomy, Foley catheter, or suprapubic aspiration) and help prevent these complications [39].

Similar to the abdominal evaluation as noted above, a low-frequency transducer is generally used for this view. With the patient supine, and with the transducer marker toward the head, the transducer is placed on the right and left flank approximately at the ninth/tenth intercostal space and angled in a slightly oblique angle to insonate between the rib spaces. The kidneys are identified on the right and left sides, caudal to the liver and spleen, respectively. To evaluate the bladder, place the transducer just superior to the pubic symphysis and aim the beam of the transducer inferiorly into the pelvis, scanning in two planes [28]. The presence of hydronephrosis is identified by a dilated collecting system at the center of the kidney, which is fluid filled and anechoic. Hydronephrosis is graded from mild to severe (Fig. 47.17). It is important to utilize the Doppler functionality to differentiate between a dilated collecting system and prominent renal vasculature, which normally produces a robust color Doppler signal.

Pelvic Pathologies

Patients with endometrial, ovarian, cervical, and metastatic cancer to the female pelvic organs may present to the emergency department with acute pelvic or abdominal pain. Adnexal masses greatly increase the risk of ovarian torsion, and POCUS may be useful in the initial screening.

The curvilinear, low-frequency transducer or endocavitary transducer is recommended for evaluating the pelvic organs. The endocavitary transducer provides better views and higher sensitivity in detecting subtle findings in early pregnancy and adnexal pathologies. B-mode is used for anatomic evaluation, and Doppler mode is necessary for assessing vascular supply and is particularly useful in cases of ovarian torsion. M-mode should be used to determine fetal heart rate as the use of Doppler carries a theoretical risk of thermal injury to a first trimester fetus and should be avoided.

Ovarian Torsion

Ovarian torsion is defined as an interruption to the ovarian vascular supply due to a rotation in the ovarian vascular ped-



Fig. 47.17 Shown here is the spectrum of hydronephrosis including mild (*left*), moderate (*middle*), and severe (*right*). The presence of hydronephrosis (*) is identified by a dilated collecting system at the center of the kidney – it is fluid filled, and anechoic, and has absent color flow on color flow Doppler. Mild hydronephrosis has dilation of

the renal pelvis only. Moderate hydronephrosis has dilation of both the renal pelvis and calyces. Severe hydronephrosis has ballooning of the renal pelvis and calyces with associated cortical thinning (*arrows*). In the image with severe hydronephrosis, there is an obstructing staghorn calculus (#)

icle which may result in ischemia and subsequent necrosis of the involved ovary [45]. Torsion begins with the twisting of the suspensory ligament which contains the ovarian artery, vein, and lymphatic vessel. Initially, the ovarian vein and lymphatic vessel are compromised which leads to ovarian edema. As the pressure increases within the ovarian capsule, arterial supply is ultimately compromised. Unfortunately, ovarian torsion is a difficult clinical entity to diagnose, as no one historical or exam feature or ultrasonographic finding can reliably exclude it [46]. As the number of sonographic signs suggestive of torsion increases, the likelihood of ovarian torsion increases [47].

Sonographic findings in ovarian torsion may include unilateral ovarian enlargement with diminished or absent Doppler flow. Other secondary signs include abnormal ovarian location, unilateral cyst or neoplasm, and free fluid in the pouch of Douglas.

The typical adult ovary measures $2 \times 2 \times 3$ cm though in cases of unilateral edema it may be many times this size [8]. In ovarian torsion, stromal edema may also lead to displacement of visible follicles to the periphery of the ovary [47]. Color Doppler and spectral Doppler of the suspected ovary should be performed to confirm the presence of both arterial and venous flow. Absent Doppler flow greatly increases the likelihood of ovarian torsion, although the presence of flow does not rule out ovarian torsion [46]. Unilateral cysts or neoplasms may be visualized as anechoic or mixed-echogenicity lesions within the relatively homogenous hypoechoic echotexture of the ovary, although their presence or absence does not always predict or exclude torsion [46, 47]. Displacement of the ovary can sometimes be noted. For example, displacement of the ovary toward the midline and superior to the uterus, where it appears posterior to the bladder as another anechoic, fluid-filled structure, is known as the "double-bladder sign" [48]. Free fluid in the pelvis, either surrounding the affected ovary or in the pouch of Douglas, in combination with a suspicious exam, may also increase the likelihood of ovarian torsion [47].

Although POCUS is a feasible first-line modality in evaluating suspected cases of ovarian torsion, it may prove challenging to the inexperienced operator, and combination of this modality with CT, MRI, or radiology ultrasound, as well as early specialist consultation, should be considered.

Deep Venous Thrombosis

The hypercoagulable state of patients with malignancy makes them more susceptible to deep venous thrombosis (DVT). Only an estimated 50% of DVT cases present with the typical manifestations of unilateral upper or lower extremity edema, erythema, and pain [49]. In a study by Casella et al., the sensitivity of clinical examination in identifying bilateral DVT was 27.2%, and the estimated incidence of having bilateral DVT in patients presenting with unilateral signs was 32% [50]. Consequently, a high index of suspicion for DVT must be maintained.

With a sensitivity and specificity greater than 95%, the mainstay of bedside diagnosis of DVT is compression ultrasonography (CUS) [51]. CUS is considered the primary diagnostic test for those who have moderate to high pretest

probability based on the Wells score or physician gestalt. The combination of CUS plus D-dimer achieves a sensitivity of 100% for DVT [52]. The typical bedside CUS exam does not evaluate for a thrombus distal to the popliteal vein (PV). Therefore, in the setting of an initial negative bedside CUS but a positive D-dimer, a repeat CUS study 1 week later is recommended to evaluate for propagation of thrombus to the deep venous system at or proximal to the PV. However, if the initial D-dimer is negative, the diagnosis of DVT is ruled out and no additional CUS is needed [49].

The high-frequency linear transducer is used with the patient supine or in reverse Trendelenburg. The leg is externally rotated and abducted with the knee slightly flexed. The transducer is oriented transverse, inferior to the inguinal ligament with the indicator toward the patient's right. To exclude thrombus, compression with the ultrasound transducer should be applied until the lumen of the vein is completely flattened with the two walls touching each other (Fig. 47.18). If the lumen cannot be fully compressed, this is diagnostic of DVT. The common femoral vein (CFV) lies adjacent to the common femoral artery (CFA). Ideally, the transducer should initially be placed at the level where the greater saphenous vein (GSV) drains into the CFV. Although the GSV is considered superficial, a thrombus in the GSV has a higher probability of propagating into the CFV and becoming a DVT. Therefore, the GSV is evaluated for compressibility at the level of the CFV, as current guidelines recommend anticoagulation for thrombus within the proximal 5 cm of the GSV (Fig. 47.19). CUS is then performed distally in a continuous fashion by compressing at 1-2 cm intervals to the level of the adductor canal when the FV penetrates deeper and becomes difficult to assess [51]. To examine the popliteal vein (PV), the scan should be performed at the popliteal fossa with the knee slightly flexed. The vein is most often located superficial to or "on top of" the popliteal artery. Caution should be exercised as the popliteal vein is a relatively deep structure, and superficial vessels, depending on body habitus, may be mistaken for the PV.

Color and spectral Doppler may be used to help differentiate veins from other structures such as arteries, cysts, or lymph nodes, which may confuse inexperienced operators, such as in Fig. 47.19c.

US is not sufficiently sensitive in diagnosing more proximal VTE within the pelvic and abdominal veins. Additionally, an evaluation for VTE distal to the popliteal vein is not typically performed in bedside CUS exams as the sensitivity of CUS in diagnosing VTE in the calf is considered poor in symptomatic and asymptomatic VTE alike



Fig. 47.18 The popliteal vein (V) seen here with an intramural thrombus (*) lies superficial to the artery (A) when examined at the popliteal fossa. The examiner is exerting pressure here to the extent that the artery is being partially compressed, but the vein (V) remains noncompressible which is consistent with DVT

[49]. If the femoral and popliteal veins are not well visualized or their compressibility is questionable, additional diagnostic tests are needed, and the study should be deemed indeterminate due to insufficient data to make or exclude the diagnosis of DVT [52].

Lymphadenopathy

Basic Approach

The linear transducer is used to image superficial lymph nodes. Color flow assessment adds critical information in the evaluation of lymph nodes.

Superficial Lymph Nodes: Benign Versus Malignant

POCUS can be used to evaluate superficial lymph nodes. Features such as size, shape, presence or absence of the hilum,



Fig. 47.19 A DVT involving the greater saphenous vein (GSV) at the junction with the common femoral vein (CFV). (a) An echogenic thrombus is seen in the lumen of the GSV with a patent anechoic CFV inferomedial to it. (b) With compression, the GSV does not compress,

whereas the lumen of the CFV is completely obliterated with compression indicating the presence of a DVT in the GSV but not the CFV. (c) Applying color Doppler shows the presence of flow within the lumen of the CFV but not the GSV where the thrombus is present

echogenicity, margins, and structural changes may be useful in distinguishing benign from malignant lymph nodes [53]. Color flow evaluation of the presence, location, and pattern of flow within the lymph node aids in this differentiation.

Size and Shape

Size and shape are lymph node characteristics reflected in the ratio between the longitudinal and the transverse or short-axis nodal measurement, termed an L/S ratio. In general, benign lymph nodes are oval, and their longitudinal axis is at least twice the transverse/short-axis measurement. Malignant lymph nodes are more often round and their L/S ratio is less than 2 [53].

Hilar Appearance

Typically, benign nodes have a central, echogenic hilum, whereas malignant lymph nodes are more likely to have absence of the hilum.

Margins

Benign nodes will have a sharp, well-defined margin, while malignant nodes may demonstrate irregular, angular, or blurred margins.

Structural Characteristics

Structural changes such as intranodal necrosis, calcification, and matting are all suggestive of malignancy. Intranodal necrosis that appears as an anechoic area of fluid within the lymph node may indicate malignancy but can also be seen in intranodal abscess or tuberculosis. Intranodal calcifications may be seen in thyroid malignancies but are rarely seen in other malignancies. Matting of lymph nodes where multiple nodes are fused in a single, poorly defined mass is seen in malignancy when there is extracapsular invasion and in infections such as tuberculosis and mononucleosis. Matting is not typically seen in reactive lymph nodes and is, therefore, useful in differentiating malignant from reactive lymph nodes. Peripheral halo and perinodal edema may be seen in tuberculosis or malignancy but are not encountered in reactive nodes [53].

Color Flow

In benign lymph nodes, a single central, vascular pedicle entering the hilum may be visualized with color flow. In contrast, malignant lymph nodes may demonstrate multiple vascular pedicles, mixed peripheral and central color flow, or overall increased color flow due to increased vascularization.

POCUS Use in Pediatric Neck Masses

In one retrospective study of children presenting with neck masses to an ED, POCUS performed by pediatric EM physicians showed 100% agreement with final diagnosis by radiology department imaging for the category of "lymph node with malignant features" [54]. All four children whose neck mass was concerning for malignant lymphadenopathy subsequently received the diagnosis of Hodgkin lymphoma. If US evaluation of the neck mass revealed a lymph node with altered architecture, loss of the central hilum, a heterogeneous echo texture, and an abnormal blood supply, it was categorized as a "lymph node with malignant features."

A case series also demonstrated that POCUS evaluation of pediatric neck swelling can identify lymphadenopathy and further characterize the lymphadenopathy as reactive lymphadenitis, abscess collection secondary to suppurative lymphadenitis, disseminated tuberculosis, and concerning for malignancy (first case with lymphadenopathy with hyperechoic foci due to calcifications, second case with enlarged lymph nodes with loss of normal echotexture and destruction of the outer capsule) [55].

A prospective pilot study involving a convenience sample of pediatric patients also suggests that pediatric EM physicians are able to use POCUS to effectively evaluate undifferentiated soft tissue neck masses in the ED. POCUS was found to be 78% accurate in the sonographic assessment of soft tissue neck masses when compared to radiology department imaging and most often diagnosed infectious or inflammatory conditions [56].

Rapid Ultrasound in Shock and Hypotension (RUSH)

Case Study

A 55-year-old female with history of metastatic breast cancer presents with altered mental status. She last received her chemotherapy via her port 5 days ago and has been ill for 1 day. Her partner reports that she awoke this morning confused and lethargic but is unable to provide further history.

The patient is ill-appearing, pale, and diaphoretic on exam. Vital signs are as follows: temperature 99.2 °F, pulse 132, SpO2 96% on room air, respiratory rate 24, and blood pressure 86/50. Her right-sided chest port is palpable. Decreased breath sounds are noted bilaterally, and tachypnea is present without accessory muscle use. Heart rate is regular and there are no murmurs heard. Abdomen is mildly distended but non-tender to exam. No extremity edema is noted.

A RUSH protocol was initiated. On evaluation of the "pump," a large pericardial effusion was noted as well as diastolic compression of the right ventricle. The "tank" evaluation demonstrated a large IVC with minimal respiratory change. Suspecting tamponade physiology, the ED team performed an ultrasound-guided pericardiocentesis, with marked improvement in the patient's blood pressure and mental status.

The RUSH protocol is one of several systematic approaches to the undifferentiated hypotensive and hypoperfused medical patient. It allows the rapid identification of several pathologies which frequently afflict oncologic patients, including cardiac tamponade, septic shock, intraabdominal hemorrhage, massive pulmonary embolism, and acute decompensated heart failure (Fig. 47.20) [3–5]. The RUSH exam integrates multiple applications as discussed above, including cardiac, thoracic, abdominal, and vascular ultrasound.

HI MAP ED Approach

The RUSH exam uses a phased-array or curvilinear transducer, and its components are captured in the "HI MAP ED" mnemonic (heart, IVC, Morison's pouch, aorta, pulmonary, ectopic, DVT). The *heart* is the first organ assessed in the RUSH exam using a phased-array transducer (or curvilinear transducer if this is not available), obtaining at least two of the four cardiac views, generally starting with the parasternal views. The heart is evaluated for the presence of pericardial effusion and, if one is present, whether there is associated tamponade (see Fig. 47.20). The LV ejection fraction is estimated or can be measured through M-mode measurement of EPSS. The right heart is assessed for signs of RV strain, including RV dilation, McConnell's sign, septal wall flattening, or decreased TAPSE, all of which raise suspicion for pulmonary embolism.

Next, the *IVC* is evaluated in the subcostal window. The presence of IVC dilation or decreased CI concurrent with pericardial effusion raises suspicion for tamponade, while its coexistence with right heart strain increases suspicion for acute obstructive shock.

The next areas to be evaluated are *Morison's* pouch in the right upper quadrant and the splenorenal space in the left upper quadrant. In oncologic patients, the frequent occurrence of ascites complicates this portion of the exam, as

ultrasound is insufficient to delineate the nature of the fluid in the abdominal cavity and cannot determine if it represents simple ascites or hemoperitoneum.

The next window of the RUSH exam is the *aortic* slide view, where the aorta is visualized in the transverse axis and assessed for evidence of aneurysm, defined as an anteroposterior diameter greater than 3 cm [57].

The *pulmonary* views are then evaluated, using the technique as described above, assessing the lungs for the presence or absence of the following pathologies: pneumothorax, pleural effusion, and B-lines or diffuse interstitial syndrome.

In female patients of child-bearing age, a pregnancy test should be checked, and the pelvic view examined for evidence of *ectopic* pregnancy. The presence of a positive pregnancy test plus free fluid in the right upper quadrant, especially in a patient without known ascites, increases the likelihood of the need for operative intervention [58, 59]. The uterus should be examined for the presence or absence of intrauterine pregnancy, which at a minimum requires the presence of a yolk sac or a fetal pole within a fluid-filled gestational sac in the uterus [8].

The presence or absence of *DVT* should be assessed with compression ultrasound of the bilateral legs. The finding of DVT plus evidence of right heart strain in a hypotensive patient may be considered diagnostic of a massive pulmonary embolus, and institutional protocols should be followed to consider the administration of thrombolytics, activation of interventional radiology, or activation of a specialized PE response team.



Fig. 47.20 Examples of pathological findings identified using the RUSH protocol. (a) Pericardial effusion seen on the subxiphoid view (*arrow*). (b) Interventricular septal deviation toward the, known as the D sign for the D shape of the LV. A sign of elevated RV pressure as may be seen in PE (*arrowheads*). (c) Minimal free fluid in the pelvic area (*arrow*) seen posterior to the uterus (*). (d) Abdominal aortic aneurysm

(*arrowhead*) measuring 6 cm in diameter (<3 cm is normal). (e) Significant IVC respiratory variation with nearly complete IVC collapse as may be seen in hypovolemia (*arrows*). (f) Multiple B-lines extending from the pleura to a depth of 18 cm concerning for diffuse interstitial syndrome. (g) A significant pleural effusion (*) superior to the diaphragm and liver (L)



Fig. 47.20 (continued)

References

- Expert Round Table on Ultrasound in ICU. International expert statement on training standards for critical care ultrasonography. Intensive Care Med. 2011;37(7):1077–83.
- Counselman FL, Beeson MS, Marco CA, Adsit SK, Harvey AL, Keehbauch JN, 2016 EM Model Review Task Force. Evolution of the model of the clinical practice of emergency medicine: 1979 to present. Acad Emerg Med. 2017;24(2):257–64.
- Keikha M, Salehi-Marzijarani M, Soldoozi Nejat R, Sheikh Motahar Vahedi H, Mirrezaie SM. Diagnostic accuracy of Rapid Ultrasound In Shock (RUSH) exam; a systematic review and meta-analysis. Bull Emerg Trauma. 2018;6(4):271–8.
- Ghane MR, Gharib M, Ebrahimi A, Saeedi M, Akbari-Kamrani M, Rezaee M, Rasouli H. Accuracy of early rapid ultrasound in shock (RUSH) examination performed by emergency physician for diagnosis of shock etiology in critically ill patients. J Emerg Trauma Shock. 2015;8(1):5–10.
- Elbaih AH, Housseini AM, Khalifa MEM. Accuracy and outcome of rapid ultrasound in shock and hypotension (RUSH) in Egyptian polytrauma patients. Chin J Traumatol. 2018;21(3):156–62.
- 6. Shriki J. Ultrasound physics. Crit Care Clin. 2014;30(1):1-24, v.
- 7. Hoskins PR, Martin K, Thrush A. Diagnostic ultrasound: physics and equipment. 3rd ed. Boca Raton: CRC Press; 2019.

- Ma OJ, Mateer JR, Blaivas M. Emergency ultrasound. 2nd ed. McGraw Hill Medical: New York; 2008.
- Burazor I, Imazio M, Markel G, Adler Y. Malignant pericardial effusion. Cardiology. 2013;124(4):224–32. https://doi. org/10.1159/000348559.
- Harada K, Ogasawara M, Shido A, Meno A, Oda S, Yoshida S, et al. Pericardial tamponade during pembrolizumab treatment in a patient with advanced lung adenocarcinoma: a case report and review of the literature. Thorac Cancer. 2020;11(5):1350–3.
- Bird BR, Swain SM. Cardiac toxicity in breast cancer survivors: review of potential cardiac problems. Clin Cancer Res. 2008;14(1):14–24.
- Ceriani E, Cogliati C. Update on bedside ultrasound diagnosis of pericardial effusion. Intern Emerg Med. 2016;11(3):477–80.
- Strony R, Linkenheil E. How to delineate pericardial effusion from pleural effusion on bedside ultrasound. J Emerg Med. 2016;50(3):495–6.
- Appleton C, Gillam L, Koulogiannis K. Cardiac Ttmponade. Cardiol Clin. 2017;35(4):525–37.
- Moore CL, Rose GA, Tayal VS, Sullivan DM, Arrowood JA, Kline JA. Determination of left ventricular function by emergency physician echocardiography of hypotensive patients. Acad Emerg Med. 2002;9(3):186–93.
- Mckaigney CJ, Krantz MJ, La Rocque CL, Hurst ND, Buchanan MS, Kendall JL. E-point septal separation: a bedside tool for emer-

gency physician assessment of left ventricular ejection fraction. Am J Emerg Med. 2014;32(6):493–7.

- Secko MA, Lazar JM, Salciccioli LA, Stone MB. Can junior emergency physicians use e-point septal separation to accurately estimate left ventricular function in acutely dyspneic patients? Acad Emerg Med. 2011;18(11):1223–6.
- Sista AK, Kuo WT, Schiebler M, Madoff DC. Stratification, imaging, and management of acute massive and submassive pulmonary embolism. Radiology. 2017;284(1):5–24.
- Pruszczyk P, Goliszek S, Lichodziejewska B, Kostrubiec M, Ciurzyński M, Kurnicka K, et al. Prognostic value of echocardiography in normotensive patients with acute pulmonary embolism. JACC Cardiovasc Imaging. 2014;7(6):553–60.
- 20. Paczyńska M, Sobieraj P, Burzyński Ł, Kostrubiec M, Wiśniewska M, Bienias P, et al. Tricuspid annulus plane systolic excursion (TAPSE) has superior predictive value compared to right ventricular to left ventricular ratio in normotensive patients with acute pulmonary embolism. Arch Med Sci. 2016;12(5): 1008–14.
- Dutta T, Frishman WH, Aronow WS. Echocardiography in the evaluation of pulmonary embolism. Cardiol Rev. 2017;25(6): 309–14.
- 22. Rudski LG, Lai WW, Afilalo J, Hua L, Handschumacher MD, Chandrasekaran K, et al. Guidelines for the echocardiographic assessment of the right heart in adults: a report from the American Society of Echocardiography endorsed by the European Association of Echocardiography, a registered branch of the European Society of Cardiology, and the Canadian Society of Echocardiography. J Am Soc Echocardiogr. 2010;23(7):685–713; quiz 786–8.
- 23. Ciozda W, Kedan I, Kehl DW, Zimmer R, Khandwalla R, Kimchi A. The efficacy of sonographic measurement of inferior vena cava diameter as an estimate of central venous pressure. Cardiovasc Ultrasound. 2016;14(1):33.
- Touw HRW, Tuinman PR, Gelissen HPMM, Lust E, Elbers PWG. Lung ultrasound: routine practice for the next generation of internists. Neth J Med. 2015;73(3):100–7.
- Lichtenstein D. Novel approaches to ultrasonography of the lung and pleural space: where are we now? Breathe (Sheff). 2017;13(2):100–11.
- 26. Lichtenstein D. Lung ultrasound in acute respiratory failure an introduction to the BLUE-protocol. Chest. 2008;134(1):117125.
- Lichtenstein DA. BLUE-protocol and FALLS-protocol: two applications of lung ultrasound in the critically ill. Chest. 2015;147(6):1659–70.
- Noble VE, Nelson B, Sutingco AN. Manual of emergency and critical care ultrasound. Cambridge, UK: Cambridge University Press; 2007.
- 29. Anderson KL, Jenq KY, Fields JM, Panebianco NL, Dean AJ. Diagnosing heart failure among acutely dyspneic patients with cardiac, inferior vena cava, and lung ultrasonography. Am J Emerg Med. 2013;31(8):1208–14.
- Wong JL, Evans SE. Bacterial pneumonia in patients with cancer: novel risk factors and management. Clin Chest Med. 2017;38(2):263–77.
- Lichtenstein D, Mezière G, Seitz J. The dynamic air bronchogram: a lung ultrasound sign of alveolar consolidation ruling out atelectasis. Chest. 2009;135(6):1421–5.
- Soni NJ, Franco R, Velez MI, Schnobrich D, Dancel R, Restrepo MI, Mayo PH. Ultrasound in the diagnosis and management of pleural effusions. J Hosp Med. 2015;10(12):811–6.
- Dickman E, Terentiev V, Likourezos A, Derman A, Haines L. Extension of the thoracic spine sign: a new sonographic marker of pleural effusion. J Ultrasound Med. 2015;34(9):1555–61.

- 34. Yang PC, Luh KT, Chang DB, Wu HD, Yu CJ, Kuo SH. Value of sonography in determining the nature of pleural effusion: analysis of 320 cases. Am J Roentgenol. 1992;159(1): 29–33.
- Mercaldi CJ, Lanes SF. Ultrasound guidance decreases complications and improves the cost of care among patients undergoing thoracentesis and paracentesis. Chest. 2013;143(2): 532–8.
- Grosu HB, Vial MR, Hernandez M, Li L, Casal RF, Eapen GA, Ost DE. Secondary spontaneous pneumothorax in cancer patients. J Thorac Dis. 2019;11(4):1495–505.
- 37. Alrajab S, Youssef AM, Akkus NI, Caldito G. Pleural ultrasonography versus chest radiography for the diagnosis of pneumothorax: review of the literature and meta-analysis. Crit Care. 2013; 17(5):1.
- Lanctôt JF, Valois M, Beaulieu Y. EGLS: Echo-guided life support. Crit Ultrasound J. 2011;3(3):123–9.
- Ledet CR, Santos D. Acute abdomen in cancer patients. In: Nates J, Price K, editors. Oncologic critical care. Cham: Springer; 2019. https://doi.org/10.1007/978-3-319-74698-2_75-1.
- Jang TB, Schindler D, Kaji AH. Bedside ultrasonography for the detection of small bowel obstruction in the emergency department. Emerg Med J. 2011;28(8):676–8.
- Ünlüer EE, Yavaşi Ö, Eroğlu O, Yilmaz C, Akarca FK. Ultrasonography by emergency medicine and radiology residents for the diagnosis of small bowel obstruction. Eur J Emerg Med. 2010;17(5):260–4.
- 42. Stengel D, Bauwens K, Rademacher G, Mutze S, Ekkernkamp A. Association between compliance with methodological standards of diagnostic research and reported test accuracy: metaanalysis of focused assessment of US for trauma. Radiology. 2005;236(1):102–11.
- 43. Cartoni C, Dragoni F, Micozzi A, Pescarmona E, Mecarocci S, Chirletti P, et al. Neutropenic enterocolitis in patients with acute leukemia: prognostic significance of bowel wall thickening detected by ultrasonography. J Clin Oncol. 2001;19(3): 756–61.
- Nesher L, Rolston KV. Neutropenic enterocolitis, a growing concern in the era of widespread use of aggressive chemotherapy. Clin Infect Dis. 2013;56(5):711–7.
- Chang HC, Bhatt S, Dogra VS. Pearls and pitfalls in diagnosis of ovarian torsion. Radiographics. 2008;28(5):1355–68.
- Robertson JJ, Long B, Koyfman A. Myths in the evaluation and management of ovarian torsion. J Emerg Med. 2017;52(4): 449–56.
- 47. Mashiach R, Melamed N, Gilad N, Ben-Shitrit G, Meizner I. Sonographic diagnosis of ovarian torsion: accuracy and predictive factors. J Ultrasound Med. 2011;30(9):1205–10.
- Lin J, Buttar S. Double bladder sign: three cases of an ultrasonographic sign that indicates ovarian torsion. Cureus. 2019;11(7):e5134.
- 49. Somarouthu B, Abbara S, Kalva SP. Diagnosing deep vein thrombosis. Postgrad Med. 2010;122(2):66–73.
- Casella IB, Bosch MA, Sabbag CR. Incidence and risk factors for bilateral deep venous thrombosis of the lower limbs. Angiology. 2009;60(1):99–103.
- Fox JC, Bertoglio KC. Emergency physician performed ultrasound for DVD evaluation. Thrombosis. 2011;2011:938709.
- 52. Pedraza García J, Valle Alonso J, Ceballos García P, Rico Rodríguez F, Aguayo López MÁ, Muñoz-Villanueva MDC. Comparison of the accuracy of emergency department-performed point-of-careultrasound (POCUS) in the diagnosis of lower-extremity deep vein thrombosis. J Emerg Med. 2018;54(5):656–64.

- Dudea SM, Lenghel M, Botar-Jid C, Vasilescu D, Duma M. Ultrasonography of superficial lymph nodes: benign vs. malignant. Med Ultrason. 2012;14(4):294–306.
- Friedman N, Tseng F, Savic R, Diallo M, Fathi K, Mclean L, Tessaro MO. Reliability of neck mass point-of-care ultrasound by pediatric emergency physicians. J Ultrasound Med. 2019;38(11):2893–900.
- 55. James V, Samuel J, Ong GY. Point-of-care ultrasound for evaluating lymphadenopathy in the pediatric emergency department: case series and review of literature. J Emerg Med. 2020;59(1):75–83.
- 56. Levine MC, Arroyo A, Likourezos A, Homel P, Dickman E. The use of point of care ultrasound in the evaluation of pedi-

atric soft tissue neck masses. Am J Emerg Med. 2019;37(8): 1466-9.

- Dent B, Kendall RJ, Boyle AA, Atkinson PR. Emergency ultrasound of the abdominal aorta by UK emergency physicians: a prospective cohort study. Emerg Med J. 2007;24(8):547–9.
- 58. Thacker JL, Reardon RF. Sonographic free fluid in patients with ectopic pregnancy. Ann Emerg Med. 2004;44(4):S85.
- 59. Moore C, Todd WM, O'Brien E, Lin H. Free fluid in Morison's pouch on bedside ultrasound predicts need for operative intervention in suspected ectopic pregnancy. Acad Emerg Med. 2007;14(8):755–8.

Part IV

Toxicities

Chemotherapy Toxicities

Katy M. Toale, Tami N. Johnson, Maggie Q. Ma, and Ngoc H. Vu

Introduction

There are many agents available to treat cancer. Chemotherapy, which includes but is not limited to alkylating agents, nitrosoureas, antimetabolites, antitumor antibiotics, topoisomerase inhibitors, and mitotic inhibitors, fights cancer through targeting cancer cells at different cell cycle phases. Other types of therapies used to treat cancer include targeted therapies, hormone therapy, and immunotherapy. Chemotherapy toxicity is a common and unfortunate consequence of therapy that can occur even at usual doses. Often these toxicities warrant emergency care. Timely recognition and appropriate management of chemotherapy toxicity in the emergency department (ED) is imperative.

While this chapter will not address all chemotherapy toxicities, we will discuss the most frequently encountered issues in the ED. Certain toxicities such as diarrhea, tumor lysis syndrome, myelosuppression, neutropenic fever, and mucositis are discussed in other chapters.

This chapter will include a review of the following chemotherapy toxicities: central nervous system toxicity, peripheral neuropathy, hypertensive crisis, nausea and vomiting, nephrotoxicity, electrolyte disorders, anaphylaxis, and extravasation.

It is important for clinicians working in the emergency setting to understand these chemotherapy toxicities and know how to manage them. Patient and caregiver education is also crucial because many of these serious side effects can occur while the patient is not hospitalized.

Case Study

BA is a 55-year-old male with bladder cancer being treated with cisplatin as part of his chemotherapy regimen. He presents to your ED complaining of weakness, fatigue, tinnitus, and paresthesias in his fingers and toes. He is able to answer questions, but his family states he has been having increasing confusion since yesterday. He has been experiencing severe nausea and vomiting and has been unable to keep any food or liquid down in the past 3 days. He is also unable to take his antiemetic medications. His vital signs are 84/65 with a heart rate of 120. His laboratory values are significant for a magnesium level of 1.1 mEq/L and sodium of 120 mEq/L and a serum creatinine of 3.5 mg/dL (baseline is 1.2 mg/dL). A computed tomography (CT) scan of the brain is unremarkable. Cisplatin is an alkylating agent with many toxicities. One of the major dose-related toxicities is nausea and vomiting which BA has been experiencing. This has caused severe dehydration and electrolyte abnormalities. In addition to nausea and vomiting, cisplatin can also cause nephrotoxicity, which manifests as hypomagnesemia and syndrome of inappropriate antidiuretic hormone (SIADH), causing hyponatremia. The tinnitus may be related to the ototoxicity of cisplatin, and the paresthesias could be related to the peripheral neuropathy associated with this chemotherapy agent. Management in the ED would include IV hydration, antiemetics, and electrolyte replacement, as well as initiation of an agent to improve the symptoms of neuropathy.

Central Nervous System (CNS) Toxicities

Chemotherapy drugs that readily cross the blood-brain barrier or administered via the intrathecal or intraventricular route can cause neurotoxicity [1]. CNS toxicity encompasses many different syndromes such as headache, somnolence, confusion, seizures, aseptic meningitis, cerebellar dysfunction, encephalopathy, intracranial hemorrhage, stroke, myelopathy, hearing loss, blindness, dementia, and coma.

K. M. Toale (🖂) · T. N. Johnson · M. Q. Ma · N. H. Vu



Department of Pharmacy, The University of Texas MD Anderson Cancer Center, Houston, TX, USA e-mail: khanzelka@mdanderson.org

CNS toxicity	Agent	
Intracranial hemorrhage	Antiangiogenic agents ^a	Asparaginase
Encephalopathy	Asparaginase Busulfan Carmustine Cisplatin Cytarabine Fluorouracil Ifosfamide	Lomustine Melphalan Methotrexate Paclitaxel Procarbazine Vincristine
Posterior reversible encephalopathy syndrome (PRES)	Antiangiogenic agents ^a Cisplatin Cytarabine Capecitabine	Gemcitabine Ifosfamide Fluorouracil Vincristine
Seizures	Busulfan Carmustine Cisplatin Cytarabine Fluorouracil Oxaliplatin	Ifosfamide Lomustine Methotrexate Paclitaxel Vincristine Any intrathecal chemotherapy agent
Cerebellar syndrome Stroke	Cytarabine Antiangiogenic agents ^a Cisplatin	Fluorouracil Methotrexate
Aseptic meningitis	Cytarabine Thiotepa Topotecan	Methotrexate Any intrathecal chemotherapy agent
Myelopathy	Cytarabine	Methotrexate
Ototoxicity	Carboplatin	Cisplatin
Blindness	Carboplatin Cytarabine	Lomustine Vincristine
Dementia	Methotrexate	

 Table 48.1
 Central nervous system toxicities and associated chemotherapy agents [1–3]

^aAntiangiogenic agents include bevacizumab, sorafenib, and sunitinib

Risk factors for developing CNS toxicity depend on the chemotherapy but may include high dosages, intrathecal or intraventricular administration, frequent administration, renal dysfunction, and hepatic dysfunction [2]. This section will focus on the more common acute neurotoxicities and their associated chemotherapy; a more extensive list of neurotoxicities caused by chemotherapy can be found in Table 48.1 [1–3].

Chemotherapy Agents Implicated

Methotrexate

Methotrexate does not easily cross the blood–brain barrier. Neurotoxicity is associated with intrathecal administration or high doses of intravenous methotrexate (>500 mg/m²) [4]. Toxicity may be due to folate or homocysteine imbalance or direct neuronal injury [5, 6]. Neurotoxicity can present in many different forms. Intrathecal methotrexate causes aseptic meningitis in greater than 10% of patients [7]. Patients typically experience headache, nausea, vomiting, nuchal rigidity, and fever. This can occur 2–4 h after methotrexate administration and last up to 12–72 h. Although this toxicity is generally self-limiting, corticosteroids such as dexamethasone can be given to prevent or treat methotrexate-induced aseptic meningitis [7].

Methotrexate neurotoxicity can also present at different times. Acute reactions occur as soon as hours after drug administration and can include aseptic meningitis as mentioned above, somnolence, seizures, stroke-like symptoms, or mental status changes [8]. Subacute neurotoxicity may present as confusion, ataxia, hemiparesis, and seizures, generally occurring 5–10 days after therapy [2]. Subacute neurotoxicity typically resolves without intervention. Lastly, chronic or delayed neurotoxicity can present months to years later, with dementia, personality changes, leukoencephalopathy, gait disturbances, aphasia, coma, and death [2, 7, 8]. These toxicities may be related to concurrent or past radiation, concomitant chemotherapy, or larger cumulative doses greater than 140 mg via the intrathecal route [2, 8]. Radiation may cause vascular injury and alter the bloodbrain barrier, allowing higher methotrexate concentrations to reach the CNS [9].

MRI imaging findings include white matter damage, diffuse white matter hyperintensity, ventricular enlargement, cortical calcifications, or cerebral atrophy [2, 8]. Leucovorin rescue therapy is usually given 24 h after methotrexate dose to decrease the toxicity associated with methotrexate folate imbalance [7]. Leucovorin 15 mg (~10–15 mg/m²) is given every 6 h for approximately 72 h, and levels may vary based on renal function and methotrexate elimination [4].

Cytarabine

Cytarabine can be given via intravenous or intrathecal administration. It is known to cross the blood-brain barrier at high intravenous doses. The primary neurotoxicity caused by cytarabine is cerebellar dysfunction (dysarthria, ataxia, nystagmus), occurring in 10-20% of patients at doses greater than 27-36 g/m², and occasionally with encephalopathy and seizures [2, 7, 8]. Cytarabine is metabolized primarily by the liver and excreted in the urine. Hepatic or renal dysfunction can lead to decreased clearance and accumulation of cytarabine [10]. Neurotoxicity is commonly seen with higher doses, in elderly patients, or in patients with renal or hepatic dysfunction [8]. The onset is generally 3-8 days after drug administration and usually resolves upon drug discontinuation but can last for weeks or months. Cytarabine also can cause loss of Purkinje cells and permanent impairment [8, 11]. MRI imaging will depict cerebellar atrophy and reversible white matter changes [2]. Other CNS toxicities that can be seen with cytarabine include blurred vision, burning eye pain, blindness, confusion, somnolence, and myelopathy [2]. Although rare, intrathecal cytarabine can cause aseptic meningitis, thought due to an immunologic response [1]. Steroid prophylaxis may prevent recurrence [10].

Ifosfamide

Up to 10–30% of patients receiving ifosfamide can experience some form of neurotoxicity, generally in the form of encephalopathy. Confusion is the most prevalent symptom, occurring in up to 80% of patients. Hallucination and psychosis can occur in up to 30% of patients, and other neurotoxicities such as lethargy, personality changes, extrapyramidal symptoms, hallucinations, seizures (including nonconvulsive seizures), and dysarthria are less common and can begin within 24 h of drug administration [2]. These toxicities are usually reversible within a few days of drug discontinuation, but cases of coma and death have occurred [12].

Risk factors for ifosfamide neurotoxicity include high doses, low albumin (potentially related to hepatic dysfunction), renal dysfunction, tumor in the lower abdomen/ pelvis, pretreatment with cisplatin, oral administration, shorter intravenous (IV) infusion time, and prior CNS disease [7, 12, 13]. It is postulated that ifosfamide neurotoxicity is due to the ifosfamide metabolite, chloroacetaldehyde. Chloroacetaldehyde crosses the blood–brain barrier and can cause direct neurotoxic effects [12, 13]. Chloroacetaldehyde also depletes glutathione stores that are needed to protect against toxicity [14].

Brain imaging for ifosfamide neurotoxicity usually shows no abnormalities, and diagnosis is generally based on exclusion of other causes [12]. Although symptoms usually resolve spontaneously, methylene blue has been reported to shorten recovery time and prevent recurrence [12, 13, 15]. An electroencephalogram should be performed in patients presenting with altered mental status to rule out nonconvulsive seizures. Patients with seizures should be treated with a benzodiazepine.

Busulfan

Busulfan readily crosses the blood–brain barrier and is commonly used in high doses in conditioning regimens for patients undergoing hematopoietic stem cell transplantation [16]. The most common neurotoxicity in patients receiving high-dose busulfan is seizures, oftentimes warranting seizure prophylaxis prior to administration of busulfan. The incidence of seizures in patients who do not receive seizure prophylaxis ranges from 5% to 15% [7]. Risk factors for seizures include older age and higher doses of 600 mg/mg² or 16 mg/kg [17, 18]. Seizure onset typically occurs within hours of busulfan administration but may occur up to 24 h after the dose is complete [7, 16]. Busulfan-induced toxicity generally presents as tonic–clonic seizures, but electroencephalography (EEG) abnormalities can be present without apparent seizures. Historically, phenytoin has been used as seizure prophylaxis in patients receiving busulfan, but caution is warranted with this agent due to its effect on busulfan metabolism. Other agents that have been recommended include levetiracetam in combination with benzodiazepines [16].

Platinum-Based Compounds (Cisplatin, Carboplatin, Oxaliplatin)

The most common form of central neurotoxicity seen with platinum-based compounds is ototoxicity (presenting as tinnitus and hearing loss), occurring in up to 20–40% of patients [19]. Ototoxicity is more common with cisplatin compared to carboplatin and oxaliplatin. Platinum-based compounds accumulate in auditory sensory cells and lead to permanent hearing loss [20]. Ototoxicity is potentially related to high cumulative dose >400 mg/m², older age, renal dysfunction, and a longer duration of therapy [21, 22]. Other forms of CNS toxicity are rare. Cisplatin administration can cause direct neurovascular toxicity manifesting as encephalopathy, cortical blindness, stroke, seizures, and focal deficits [2, 20, 23]. Concomitant use of cisplatin with other chemotherapy agents or electrolyte abnormalities can increase the risk of neurotoxicity [1, 24].

Diagnosis and Treatment

Chemotherapy-induced neurotoxicity is a diagnosis of exclusion. Other causes for neurologic changes should be ruled out and treated. The treatment of chemotherapy-induced neurotoxicity generally involves prompt discontinuation of the offending agent. Depending on the agent, rechallenge with dose reductions and/or longer intervals between cycles may be considered [2]. In many cases, neurotoxicity is reversible upon discontinuation though it may take months for recovery; however, it can be irreversible and lead to permanent CNS damage.

Peripheral Neuropathy

Chemotherapy-induced peripheral neuropathy (CIPN) can significantly worsen quality of life and have an effect on cancer treatment due to dose reductions, treatment delays, and/or discontinuations [25]. In addition, this toxicity can also result in excessive healthcare costs and resources utilized [26]. Certain factors place patients at higher risk for nerve damage. These include the type of chemotherapy used, duration of treatment, cumulative dose, age, diabetes, African race, folate/vitamin B12 deficiency, smoking, impaired renal function, cancer-associated neuropathy, and prior or concomitant use of neurotoxic agents [25, 27].

Pathophysiology

The exact cause of CIPN differs between classes of chemotherapy agents. For platinum-based chemotherapy, it is thought the antitumor mechanism causes changes in the function of neuronal and glial cells. Mitochondria dysfunction can result in neuroinflammation, DNA damage, and axonal degeneration. Mechanisms for taxane-induced CIPN include axonal microtubule disruption, mitochondrial dysfunction, axon degeneration, altered calcium homeostasis, changes in peripheral nerve excitability, and activation of the immune system with development of neuroinflammation. Vinca alkaloids cause changes to large axons and dorsal root ganglion (DRG) neurons, leading to hyperexcitability of peripheral neurons and activation of immune cells, which leads to neuroinflammation. Bortezomib causes mitochondrial damage, increases in sphingolipid metabolism within astrocytes, and activation of immune cells, which results in neuroinflammation. Potential mechanisms for thalidomide-induced CIPN include an antiangiogenic effect, which may cause damage to sensory neurons and downregulation of tumor necrosis factor alpha (TNF- α) levels and inhibition of nuclear factor kappa B (NF-kB), which accelerates neuronal cell death. Ixabepilone-induced CIPN is postulated to cause microtubule disruption, damage of mitochondria, and activation of immune cells, resulting in neuroinflammation [27].

Epidemiology

Estimates vary but according to a recent review, 68% of patients develop CIPN after the first month of therapy, 60% at 3 months and 30% after 6 months [28]. There are many assessment methods for CIPN, but no single method is considered superior. This leads to a wide range in the reported prevalence of CIPN [29]. A detailed list of the most common agents implicated, incidence, and risk factors can be seen in Table 48.2 [25, 27, 30].

Presentation/Diagnosis

Peripheral neuropathy is clinically defined as any form of damage, inflammation, or degeneration of peripheral nerves. Patients may experience sensory nerve damage in addition to motor and autonomic nervous system damage. Symptoms usually present within the first 2 months of chemotherapy treatment and can progress until treatment is discontinued [27, 30]:

- Most often, symptoms are sensory, including paresthesias and pain.
- Symptoms may occur at any time during treatment, even after treatment is stopped.
- Symptoms are most often symmetrical.

Chemotherapy			
agent	Incidence	Additional risk factors	Others
Taxanes Paclitaxel Docetaxel	11–87% More common Less common	Cumulative dose Duration of infusion Simultaneous administration of platinum-based compounds History of peripheral neuropathy	Symptoms include paresthesia, numbness, or neuropathic pain in the hands and feet Symptoms usually improve after discontinuation but can continue indefinitely
Platinum- based compounds Cisplatin Oxaliplatin	70–100%	Cumulative dose Longer exposure times Oxaliplatin only: Cold temperatures Time of infusion Low body weight Preexisting peripheral neuropathy	Symptoms include numbness, tingling, and paresthesia in the hands and feet Symptoms usually improve or resolve within a year of completion of therapy; however, in some cases, the damage is not reversible "Coasting" phenomenon possible with worsening symptoms following discontinuation
Vincristine	60%	Cumulative dose	Symptoms include pain in hands and feet, muscle weakness, and cramping Symptoms are reversible upon discontinuation of therapy
Bortezomib	34%	Higher dosages Less toxicity with subcutaneous administration	Symptoms include chronic sensory peripheral neuropathy often with a neuropathic pain syndrome Occurs early in treatment and may last for weeks, months, or even years after discontinuation
Thalidomide	20-60%	Cumulative dose Advanced age Prior neuropathy	Symptoms include paresthesias, numbress and loss of dexterity on the toes and fingers
Ixabepilone	60–65%	Cumulative dose Prior doses of chemotherapy	Symptoms include paresthesia, numbness, or neuropathic pain in the hands and feet Symptoms usually improve after discontinuation

Table 48.2 Characteristics of chemotherapy agents that cause chemotherapy-induced peripheral neuropathy [25, 27, 30]

- Symptoms begin in the fingers and toes and spread proximally.
- In severe cases, loss of sensory perception can occur.

Prevention

Unfortunately, no agents have been proven beneficial for the prevention of CIPN [30]. Venlafaxine demonstrates some benefit for prevention of oxaliplatin CIPN; however, it is not routinely used due to concerns regarding its anticancer activity [31]. The clinician can consider dose reduction or discontinuation of the offending agent for patients unable to tolerate CIPN symptoms [25].

Management

Several agents have been tested for efficacy in treating CIPN. The most commonly used treatments for CIPN include topical amitriptyline/ketamine/baclofen, gabapentin, pregabalin, and duloxetine. Tricyclic antidepressants are commonly used in other neuropathic pain syndromes but have shown no benefit for CIPN, and adverse effects are a limitation. Characteristics of each medication can be seen in Table 48.3 [32]. Unfortunately, there is no gold standard for CIPN detection and grading, making clinical trials difficult to compare. Recently, guidelines on designing trials for prevention of CIPN have been published [33].

The agent with the highest level of recommendation for treatment of CIPN is duloxetine. Initial pilot studies utilizing duloxetine showed promise [34, 35]. Duloxetine was then evaluated in a randomized, doubleblind, placebo-controlled crossover trial of 231 patients with CIPN from paclitaxel or oxaliplatin. Patients in the duloxetine group experienced less pain as well as a greater decrease in the pain that interfered with daily functions. The patients with oxaliplatin-induced peripheral neuropathy had better outcomes than patients on a taxane, and this may be explained by the different mechanisms causing CIPN [36]. Another ongoing trial should provide additional information to guide duloxetine use in CIPN [37].

In a study evaluating topical baclofen 10 mg, amitriptyline 40 mg, and ketamine 20 mg in pluronic lecithin organogel vs. placebo for CIPN, the topical compound showed a trend toward improved outcomes with no toxicities reported [38].

Gabapentin and pregabalin are used frequently for many types of neuropathic pain; however, current data does not support their use in CIPN. A trial of 115 patients who were randomly assigned to gabapentin or placebo showed no difference in outcomes [39]. Pregabalin was shown to improve outcomes in smaller trials [40]; however, a phase 4 study was terminated early when an interim analysis failed to detect a difference in outcomes [41].

Acupuncture has been studied for CIPN, and at this time there is not enough evidence to recommend it; however, it is considered safe and could potentially benefit patients suffering from CIPN [42]. Occupational and physical therapy trials are also limited but offer potential benefits to issues commonly faced by patients with CIPN, including gait issues, impaired postural control, and altered sensory organization [43].

Agent	Dose	MOA	Adverse effects	Comments
Duloxetine	30 mg PO daily for 1 week and then increase to 60 mg PO daily	Potent inhibitor of neuronal serotonin and norepinephrine reuptake and a weak inhibitor of dopamine reuptake	Fatigue Insomnia	Best evidence to support use in CIPN
Pregabalin	150 mg daily initially and then increase to 300– 600 mg in two divided doses	Modulates calcium channel activity by binding to the $\alpha 2\delta$ receptor site	Somnolence Dizziness Peripheral edema Xerostomia Ataxia Weight gain	Schedule V controlled substance Do not discontinue abruptly
Gabapentin	300–900 mg daily initially and then increase to 1200–3600 mg in three divided doses	Modulates calcium channel activity by binding to the $\alpha 2\delta$ receptor site	Somnolence Dizziness Ataxia Confusion Disorientation	Do not discontinue abruptly
Topical amitriptyline/ ketamine/baclofen	Apply two to three times daily	Amitriptyline: increases the synaptic concentration of serotonin and/or norepinephrine in the CNS by inhibition of their reuptake Ketamine: noncompetitive NMDA receptor antagonist that blocks glutamate	Anticholinergic effects Postural hypotension Sedation	Formulations vary and can include other agents such as lidocaine

Table 48.3 Characteristics of agents used in the treatment of chemotherapy-induced peripheral neuropathy (CIPN) [32]

CNS central nervous system, MOA mechanism of action, NMDA N-methyl-D-aspartate receptor, PO by mouth

Hypertensive Crisis

Hypertension is common in the general population and can lead to myocardial infarction, stroke, renal failure, and death if not detected early and treated appropriately [44]. Hypertension is a known long-term comorbidity of many patients with malignancy. The prevalence before treatment is similar to the community but increases after the initiation of chemotherapy and targeted agents [45]. The incidence of hypertension depends on age, cardiovascular comorbidities, type of malignancy, treatment regimen, and concomitant medications [46]. The estimated risk in patients with malignancy is as high as 36% [47].

Pathophysiology

The mechanisms underlying worsening hypertension in patients with malignancy are not well understood. It is hypothesized that certain chemotherapy and targeted or supportive agents associated with hypertension contribute to endothelial dysfunction with nitric oxide reduction, increase in vascular tone, decrease density of microvessels, renal thrombotic microangiopathy that leads to proteinuria, arterial vasoconstriction, sodium and fluid retention, and activation of the renin–angiotensin system [46].

Clinical Features

As in the general population, patients with malignancy are at risk for hypertensive crisis. For all levels of hypertension severity, including crisis level, Fraeman and colleagues reported that the incidence was considerably higher during periods of chemotherapy exposure versus without exposure. Patients experiencing crisis-level hypertension rates during exposure versus without, respectively, were 8.98 and 2.09 cases per 100 person-years [47]. Hypertensive emergencies are always associated with end-organ damage, such as hypertensive encephalopathy, cerebral infarction or hemorrhage, myocardial ischemia or infarction, heart failure, aortic dissection, and/or renal failure. Hypertensive emergency is not related to any specific blood pressure (BP) number but usually involves an acute elevation of the systolic BP (SBP) greater than 180 mm Hg or diastolic BP (DBP) above 120 mm Hg. The pace of the rise and percent increase in BP as well as the presence of end-organ damage is far more important than the actual number and demands immediate interventions to prevent further end-organ damage [48, 49].

Chemotherapy and Targeted Agents Implicated

The higher rate of hypertension in patients with malignancy can be contributed to the use of chemotherapy and targeted agents that can cause hypertension (angiogenesis inhibitors, 17-80%; alkylating agents, 36-39%; and immunosuppressants after stem cell transplantation, 30–80%) [45, 46]. Disruption of vascular endothelial growth factor (VEGF) signaling using antibody against VEGF-A (bevacizumab), antibody against receptor for VEGF (VEGFR2) (ramucirumab), and small molecule inhibitors of receptors for VEGF (sunitinib, sorafenib, etc.) is associated with hypertension, and its occurrence may be a useful marker of efficacy for these agents [50]. Some of the end-organ effects (e.g., heart failure, stroke, or renal failure) of these chemotherapy agents can compound hypertension or cause primary hypertensive crisis. Over 150 chemotherapy and targeted agents have been associated with hypertension, heart failure, stroke, and/or renal failure [32, 46].

Treatment

There is no specific guideline for treating hypertension or specifying which agents to use in patients with malignancy. Patients with hypertensive urgency should be treated, and this is usually achieved by administering oral agents followed by several hours of clinical observation [49, 51] (see Table 48.4 [32, 48, 52]).

Aggressive blood pressure control is mandatory for patients with hypertensive emergency in order to minimize the risk of end-organ damage. Patients with hypertensive emergency require admission to an intensive care unit (ICU), intravenous (IV) antihypertensive agents, and continuous BP monitoring. Based on the Seventh Report of the Joint National Committee (JNC-7) guidelines, the generalized BP goal is to lower the mean arterial pressure (MAP) by 20-25% within the first hour while avoiding excessive decreases in BP. When the patient is stable, the SBP can be lowered to 160 mm Hg, and DBP can be lowered to 100-110 mm Hg within the next 2-6 h. If this level of BP control is tolerated and the patient is stable, a gradual reduction to the patient's baseline BP can be achieved over the next 24-48 h [48, 49, 53]. There are exceptions to the generalized BP goal for patients who have hypertensive emergencies. This includes patients with certain complications such as acute aortic dissection, acute intracerebral hemorrhage, and acute ischemic stroke, with or without reperfusion therapy. The specific blood pressure goals are listed in Table 48.5 [48, 49, 52, 54, 55].

Agent	MOA	Dose	Onset	Duration	Common adverse effects ^a
Captopril	ACE inhibitor	12.5–50 mg PO every 1–2 h	15–30 min	4–6 h	Acute renal failure Angioedema Cough
Clonidine	Central α_2 -agonist	0.1–0.2 mg PO every 1–2 h	30–60 min	6–8 h	Bradycardia Dry mouth Rebound hypertension after withdrawal Sedation
Furosemide	Loop diuretic	20–40 mg PO every 2–3 h	30–60 min	8–12 h	Hypokalemia Hyponatremia Volume depletion
Labetalol	$A_1, \beta[beta]_{1\&2}\text{-blocker}$	200–400 mg PO every 2–3 h	30–120 min	6–8 h	Bronchoconstriction Heart block Heart failure Hypotension Vomiting
Nitroglycerin 2%	Nitrate	1–2 inches topically every 6 h	20–60 min	4–8 h	Bradycardia Dizziness Headache Methemoglobinemia

 Table 48.4
 Common antihypertensive agents used for hypertensive urgencies [32, 48, 52]

ACE angiotensin-converting enzyme, MOA mechanism of action, PO per os, by mouth

^aHypotension can occur with all agents

Table 48.5 Agents for treating hypertensive emergencies with comorbidities and blood pressure goals [48, 49, 52, 54, 55]

Comorbidity	Preferred agent(s) ^a	Blood pressure goal
Acute aortic dissection	Esmolol ^b	SBP <120 mm Hg and heart rate <60 beats/minute within 20 min; lowest BP possible that maintains end-organ perfusion (SBP 70–90 mm Hg)
Acute heart failure (pulmonary edema)	Loop diuretics Nitroglycerin Nitroprusside	Generalized goal ^e
Acute ischemic stroke	Clevidipine Labetalol	Ineligible for reperfusion therapy: <220/120 mm Hg, decrease no more than 15%
	Nicardipine	Eligible for reperfusion therapy: <185/110 mm Hg
	Nitroprusside	During and post-reperfusion therapy for at least 24 h: <180/105 mm Hg
		Planned mechanical thrombectomy and have not received IV fibrinolytic therapy: <185/110
		During and post mechanical thrombectomy for 24 h: <180/105 mm Hg
Acute intracerebral hemorrhage	Clevidipine	SBP 150-220 mm Hg: Acute lowering of SBP to140 mm Hg is probably safe
	Labetalol Nicardipine Nitroprusside	SBP >220 mm Hg: May be reasonable to consider aggressive reduction
Acute myocardial infarction	Clevidipine ^d Esmolol/labetalol Nicardipine ^d Nitroglycerin	Generalized goal ^e
Acute renal failure	Clevidipine Fenoldopam Nicardipine	Generalized goal ^c

BP blood pressure, *CPP* cerebral perfusion pressure, *ICP* intracranial pressure, *MAP* mean arterial pressure, *mm* Hg millimeters of mercury, *SBP* systolic blood pressure

^aAgents are listed in alphabetical order, not in preference

 ${}^{\boldsymbol{b}}\boldsymbol{\beta}[\text{beta}]\text{-blockade must be used prior to other vasodilators}$

^cDecrease MAP by 20–25% during first hour; if patient is stable, decrease SBP to 160 mm Hg and DBP to 100 mm Hg over next 2 to 6 h, then a gradual reduction to the patient's baseline BP over the next 24 to 48 h

^dMay be used in patients with heart rate <70 beats/min

Agent	MOA	Bolus dose	Continuous infusion dose	Dose titration	Onset	Duration	Adverse effects ^a
Clevidipine ^b	Calcium channel blocker	Not applicable	1–16 mg/h	1 mg/h every 2 min	2–4 min	5–15 min	Flushing Headache Heart failure Tachycardia
Esmolol	β[beta] 1-blocker	250–500 mcg over 1 min; may repeat after 5 min	25–300 mcg/kg/min	25 mcg/kg/ min every 5 min	1–2 min	10–30 min	Bronchoconstriction Heart block Heart failure
Fenoldopam ^c	Dopamine-1 receptor agonist	Not applicable	0.1–1.6 mcg/kg/min	0.05–0.1 mcg/ kg/min every 15 min	5–10 min	10–15 min	Flushing Headache Tachycardia
Labetalol	$\alpha[alpha]_1,$ $\beta[beta]_{1\&2}$ -blocker	20–80 mg over 2 min every 10 min	2–8 mg/min; max dose for bolus and infusion is 300 mg/24 h	Titrate by 0.5 mg/min every 60 min	5–10 min	2–6 h	Bronchoconstriction Heart block Heart failure Vomiting
Nicardipine	Calcium channel blocker	Not applicable	5–15 mg/h	2.5 mg/h every 5 min	5–10 min	2–4 h	Flushing Headache Reflex tachycardia
Nitroglycerin	Converts to NO and increases cGMP	Not applicable	5–200 mcg/min	5–10 mcg/ min every 5 min	1–3 min	5–15 min	Headache Methemoglobinemia Tachycardia Vomiting
Sodium nitroprusside ^d	Increase cGMP, blocks intracellular calcium	Not applicable	0.25–10 mcg/kg/min	Titrate by 0.5 mcg/kg/ min every 5 min	Immediate	1–2 min	Cerebral auto regulation impairment Coronary steal Cyanate toxicity Nausea/vomiting

 Table 48.6
 Common parenteral agents used for hypertensive emergencies [32, 48, 49]

cGMP cyclic guanosine monophosphate, h hour, kg kilogram, mcg micrograms, mg milligrams, min minute, MOA mechanism of action, NO nitric oxide

^aHypotension may occur with all agents

^bCaution with severe aortic stenosis and acute heart failure

°Caution with glaucoma and sulfite allergies

^dCaution with high intracranial pressure, hepatic and renal failure

Treatment decisions with parental agents should focus on the end-organ system at risk and associated side effects caused by the specific agent used. See Table 48.5 [48, 49, 52, 54, 55] and Table 48.6 [32, 48, 49] for recommended intravenous agents per end-organ system.

Aortic Dissection Acute aortic dissection has a high mortality rate of 1% per hour over the first several hours [56]. Early diagnosis and treatment is crucial for survival. Intravenous, fast-acting β -blockers are the drug of choice due to their ability to lower the heart rate and reduce aortic stress [57].

Heart Failure Patients with heart failure, severe hypertension, and significant fluid overload should be treated initially with intravenous loop diuretics to reduce mortality [58]. Nitroglycerin and sodium nitroprusside are the most commonly used antihypertensive agents to help with filling pressures and left ventricular afterload [58].

Ischemic Stroke Appropriate management for ischemic stroke is very important. Studies have demonstrated a

"U-shaped" relationship between BP and clinical outcome [59]. The concern with lowering the BP in patients with ischemic stroke is expansion of the central ischemic core. Cerebral autoregulation can be lost after a stroke, leaving the core, and the surrounding ischemic penumbra, prone to hypoperfusion and a potentially worse outcome [60]. There are limited studies evaluating the optimal BP goal and appropriate pharmacotherapy. Modest lowering of the BP is recommended and dependent on the patient's eligibility for reperfusion therapy [52].

Several small studies have compared labetalol to nicardipine for lowering BP. In a meta-analysis, these two agents had comparable side effects and efficacy for the treatment of hypertension in patients with several different types of strokes, including acute ischemic stroke, intracerebral hemorrhage, and subarachnoid hemorrhage [61]. In a more recent study, intravenous labetalol and nicardipine continuous infusions were comparable with no significant difference in time to BP goal, variability in BP, and use of rescue antihypertensive medications [62]. **Hemorrhagic Stroke** The most acute concern after hemorrhagic stroke is hematoma volume expansion [63]. Hematoma expansion occurs very early (first 3 h), with limited expansion beyond 24 h [59]. Two studies have confirmed the feasibility and safety of early rapid BP lowering in patients with intracerebral hemorrhage (Intensive Blood Pressure Reduction in Acute Cerebral Hemorrhage Trial [INTERACT] pilot study and the Antihypertensive Treatment in Acute Cerebral Hemorrhage [ATACH] trial) [64, 65]. Labetalol and nicardipine are used to control BP in patients with hemorrhagic stroke as mentioned in the ischemic stroke section. Clevidipine is also used to lower BP in patients with a hemorrhagic stroke. No difference in safety and efficacy was found when compared to nicardipine [66–68].

Acute Coronary Syndromes The goal for patients experiencing acute coronary syndrome is to decrease myocardial oxygen demand and improve coronary perfusion. For patients experiencing hypertensive emergencies, both nitrates and β -blockers have been used to reach the above goal [69].

Renal Failure Fenoldopam has an indication for hypertensive emergencies and improves creatinine clearance, urine flow rates, and sodium excretion in severely hypertensive patients with both normal and impaired renal function [49, 66]. Therefore, it may be useful in patients with hypertensive emergencies and renal failure. Diuretic use in patients with severe hypertension and renal failure may be either beneficial or harmful and is completely dependent on the patient's volume status.

It is absolutely crucial to recognize hypertensive emergencies as soon as possible. End-organ damage can be minimized by early diagnosis and treatment. Treatment and the BP goal are dependent on the end-organ involved.

Nausea and Vomiting

Chemotherapy-induced nausea and/or vomiting (CINV) is a debilitating side effect of chemotherapy, potentially impacting fluid and electrolyte balance, nutrition status, future treatment courses, quality of life, and overall cost of healthcare. Direct healthcare costs of treating CINV per day can range from US\$1300 to US\$2400, not accounting for indirect costs associated with lost workdays or productivity losses [70]. After the introduction of serotonin (5-HT₃) receptor antagonists, prophylactic antiemetic regimens were able to reduce the incidence of CINV. Despite the improvement, CINV refractory to these prophylactic regimens occurs in up to 40% of patients receiving chemotherapy [71]. CINV can

be characterized as acute, delayed, breakthrough, refractory, or anticipatory. This section will focus on management strategies for breakthrough CINV only.

Pathophysiology

CINV is initiated by the peripheral stimulation of enterochromaffin cells in the GI tract and the central stimulation of the chemoreceptor trigger zone by chemotherapeutic agents. Centrally, substance P release, which is triggered by chemotherapy, binds to NK-1 receptors and stimulates the chemotherapeutic trigger zone. Peripherally, stimulation of enterochromaffin cells causes a release of serotonin (5-HT), which sends information to the central nervous system [71, 72]. These signals are received and processed by the vomiting center in the medulla oblongata, which are then converted to signals promoting emesis [73, 74]. Serotonin is the predominant peripheral neurotransmitter involved in the first 24 h of CINV, while signals from dopamine and substance P (which corresponds with the primarily centrally located neurokinin-1 receptor) predominate after 24 h [75]. Other neurotransmitters and receptors involved include acetylcholine, corticosteroid, cannabinoid, histamine, neurokinin, and opioid [76].

Clinical Features

Nausea is an unpleasant sensation in the back of the throat or epigastrium that may or may not result in emesis. Vomiting is a motor reflex, which results in a forceful expulsion of GI contents [73, 74]. CINV can be categorized into acute and delayed phases. Acute CINV occurs within the first 24 h of chemotherapy, while delayed CINV occurs after 24 h and can last up to 7 days. Breakthrough CINV is any nausea or vomiting that occurs despite optimal prophylactic antiemetic regimens [76].

Chemotherapy Agents Implicated

Chemotherapeutic agents are classified into four different categories based on their emetogenicity risk. Highly emetogenic agents have a risk of greater than 90% (meaning 90% or more of patients receiving these agents will experience emesis without prophylactic antiemetics), moderately emetogenic agents have a risk of 30–90%, agents with low emetogenicity have a risk of 10–30%, and minimally emetogenic agents have a risk of less than 10% for emesis [77]. Other risk factors for CINV include female sex, younger age, history of little to no alcohol use, history of CINV or morning sickness during pregnancy, and susceptibility to motion sickness. A chart grouping chemotherapy agents by emetogenicity risk can be found in the National Comprehensive Cancer Network (NCCN) guidelines [76].

Treatment

Treatment of breakthrough CINV is more challenging than preventing CINV. There are currently three published guidelines from three different cancer organizations-NCCN, American Society of Clinical Oncologic (ASCO), and Multinational Association of Supportive Care in Cancer/ European Society for Medical Oncologic (MASCC/ESMO) [76, 78, 79]. NCCN provides the most comprehensive recommendations for breakthrough CINV. NCCN recommends using an agent different from those in the prophylactic regimen and suggested that multiple agents with different mechanisms of action may be needed. There is also a recommendation to utilize these agents on a scheduled, around-theclock dosing strategy versus an as-needed basis. Therapeutic options include benzodiazepines, dopamine antagonists (including phenothiazines, olanzapine, haloperidol, and metoclopramide), cannabinoids, antihistamines/anticholinergics, corticosteroids, and serotonin receptor subtype 3 $(5-HT_3)$ antagonists. Additionally, it may be beneficial to initiate an H₂ antagonist or proton pump inhibitor to treat concurrent dyspepsia, as well as maintain optimal hydration status and electrolyte balance [76]. See Table 48.7 [72, 73, 75, 76, 80-82] for details on pharmacotherapeutic treatment options for CINV. Beyond a small, phase 2 trial [83], neurokinin 1 (NK-1) receptor antagonists such as aprepitant have been studied primarily as part of prophylactic regimens and have not been recommended by any of the three main antiemesis guidelines from ASCO, NCCN, or MASCC/ESMO for the treatment of breakthrough CINV [76, 78, 79]. Aside from guideline recommendations, considerations such as previous success with certain antiemetic classes, drug-drug interactions, and patient-specific characteristics (e.g., allergies or QTc interval) should all be taken into account when selecting an antiemetic regimen for breakthrough CINV.

There is currently a paucity of literature to guide the practitioner in selecting antiemetics for breakthrough CINV. The majority of recent data highlights the antipsychotic, olanzapine. Likely due to its effect on multiple receptor sites involved in the pathophysiology of CINV, olanzapine has shown to be beneficial in treating breakthrough CINV [76].

One retrospective study observed 33 patients who experienced refractory CINV that failed to respond to both benzodiazepines and dopamine antagonists and received at least one dose of olanzapine. Of these patients, 65–70% had success with olanzapine. The typical dose given was 5–10 mg by mouth daily for a median of 4 days [75]. Another observational study evaluated medications received for breakthrough CINV. Of 39 patients who required rescue antiemetics, 88% received prochlorperazine, while 12% received a 5-HT₃ antagonist. Both groups reported a 75% reduction of nausea after 4 h. Both groups also noted significant symptom control within 30 min [84]. In one of the only randomized controlled trials evaluating treatment for CINV, Navari et al. studied 108 patients receiving either olanzapine or metoclopramide for breakthrough CINV. Patients were included if they received highly emetogenic chemotherapy with appropriate prophylactic antiemetic regimens. Olanzapine was given as a 10 mg dose orally every 24 h for 72 h, while metoclopramide was given as a 10 mg dose orally every 8 h for 72 h. Patients receiving olanzapine had a significantly lower incidence of both nausea and vomiting, and olanzapine was well tolerated [85]. In a retrospective study by Chiu et al., olanzapine was evaluated in 193 cases of breakthrough CINV and was found to improve nausea 88.1% of the time and vomiting 21.8% of the time. Several adverse effects were reported including sedation (42% of cases) and constipation (32% of cases). Of note, the majority of patients in this trial (94%) received a lower dose of olanzapine at 2.5 mg daily [86]. Lastly, Nakagaki et al. randomized 62 stem cell transplant patients (following chemotherapy) to ondansetron 32 mg infused over 24 h, olanzapine 10 mg daily + ondansetron IV three times daily, or a single dose of palonosetron 0.25 mg IV. The primary composite outcome of no emesis, no rescue medication use, and improvement in nausea by $\geq 50\%$ was significantly higher in the olanzapine + ondansetron group at 24 h (45% versus 6% in the ondansetron infusion group and 18% in the palonosetron group) and at 48 h (64% versus 6% in the ondansetron infusion group and 18% in the palonosetron group) [87]. With growing evidence supporting olanzapine use in the treatment of breakthrough CINV, NCCN recommends treatment of CINV with olanzapine as a category 1 recommendation (unless olanzapine was used as part of the prophylactic regimen) [76].

CINV continues to be a significant adverse effect of chemotherapy that can impact the treatment of cancer. Utilizing the most effective prophylactic regimens based on the emetogenicity of the chemotherapy agent(s) is most important, but patients may still develop breakthrough CINV. In these cases, it is important to quickly control symptoms with scheduled antiemetics that have mechanisms differing from initial agents used for prophylaxis.

Nephrotoxicity

Chemotherapy-induced nephrotoxicity (CIN) is a significant complication of therapy, resulting in treatment delays, dosage reductions, and therapy discontinuation. In 2015, the International Serious Adverse Event Consortium report classified drug-induced kidney disease into four main

Agent	Dose	MOA	Adverse effects	Comments
5-HT ₃ Receptor A	ntagonists			
Ondansetron	16–24 mg PO daily or 8–16 mg IV	Antagonistic effect at the 5-HT ₃ receptor located in the GI tract, CTZ and vomiting	QTc prolongation	PO and IV are equally effective All are equally effective/safe
Dolasetron Granisetron	100 mg PO daily 1–2 mg PO daily or 1 mg PO twice daily or 0.01 mg/ kg (maximum 1 mg) IV daily or 3.1 mg/24-h TD patch every 7 days	center	Headache Constipation	when given in biologically equal doses If palonosetron was used in the prophylactic regimen, a different mechanism should be targeted for breakthrough CINV Most benefit seen when administered around-the-clock (scheduled) PO ondansetron has less QTc prolongation risk than IV
Dopamine antago	nists			
Haloperidol	0.5–2 mg PO/IV every 4–6 h	Butyrophenone, most potent dopamine antagonist; weak anticholinergic and alpha-1 adrenergic blocking effects	Sedation QT _c prolongation Dystonic reactions Extrapyramidal symptoms	Antiemetic properties seen at doses lower than antipsychotic doses
Metoclopramide	10–20 mg PO/IV every 4–6 h	Benzamide analog, peripheral dopamine antagonist; stimulates prokinesis via serotonin (5-HT ₄) receptors	Dystonic reactions Tardive dyskinesia Akathisia Diarrhea Mild sedation Orthostatic hypotension	Increased efficacy at higher doses due to additional serotonin blockade Tardive dyskinesia risk is cumulative dose-dependent and more likely to occur with chronic use
Prochlorperazine	10 mg PO/IV every 6 h (max 40 mg/day) or 25 mg PR twice daily	Phenothiazine; dopamine antagonist; some anticholinergic and alpha-adrenergic blocking effects; prochlorperazine with	Sedation Dystonic reactions	Due to increased histamine receptor antagonism, promethazine causes more
Promethazine	12.5–25 mg PO/IV every 4–6 h or 25 mg PR every 6 h	more predominant dopamine antagonism; promethazine with more antihistamine properties	Extrapyramidal symptoms	sedation than prochlorperazine
Olanzapine	5–10 mg PO daily	Effects on multiple receptors (serotonin, dopamine, acetylcholine, muscarinic, histamine, alpha-1 adrenergic)	Sedation Orthostatic hypotension Increased appetite Dystonic reactions	Administer at bedtime to avoid daytime sedation Consider dose reduction (to 5 or 2.5 mg if higher dose causes excessive sedation) Sedation improves after day 2
Cannabinoids				
Dronabinol	5–10 mg capsule PO every 6–8 h	Effects on the cannabinoid receptors in the CNS and peripheral receptors	Sedation Dizziness	Use is limited by side effects Titrate dose upward to
Nabilone	1–2 mg PO twice daily		Dysphoria Postural hypotension Hallucinations Appetite stimulation	minimize side effects Dronabinol oral solution has greater bioavailability than the capsule (2.1 mg oral solution = 2.5 mg capsule)
Benzodiazepines	0.5.0 00/01		0.1.4	
Lorazepam	0.5–2 mg PO/IV every 6 h	Binds to benzodiazepine receptors in the postsynaptic GABA receptors	Sedation Respiratory depression Hypnosis/ amnesia	Added benefit of reducing anxiety Not recommended as monotherapy Caution when used with opioids

Table 48.7 Pharmacologic treatment options for breakthrough chemotherapy-induced nausea and vomiting (CINV) [72, 73, 75, 76, 80-82]

(continued)

 Table 48.7 (continued)

Agent	Dose	MOA	Adverse effects	Comments
Corticosteroids				
Dexamethasone	12 mg PO/IV daily	Not fully understood	Insomnia Increased appetite Hyperglycemia GI distress	Is more commonly used in prophylactic regimens for CINV Administer in the morning to alleviate insomnia side effect
Other Agents				
Scopolamine patch	1.5 mg TD every 72 h	Antimuscarinic, anticholinergic	Sedation Fatigue Paradoxical CNS excitation Hallucinations Xerostomia	May benefit if CINV caused by excessive secretions or includes component of motion sickness

CNS central nervous system, CTZ chemotherapy trigger zone, GABA gamma-aminobutyric acid, GI gastrointestinal, IV intravenous, kg kilogram, mg milligrams, MOA mechanism of action, PO by mouth, PR per rectum or rectally, TD transdermal

types: acute kidney injury (AKI), glomerular disorder, tubular disorder, or nephrolithiasis/crystalluria. AKI is further broken down to acute tubular necrosis (ATN), acute interstitial nephritis (AIN), and osmotic nephrosis [88]. Chemotherapy can affect any area of the kidney including the glomerulus, tubules, interstitium, or microvasculature. Adverse effects may range from a simple rise in serum creatinine without other sequelae to acute kidney failure requiring dialysis [89].

Risk Factors

Several factors increase the risk of CIN including intravascular volume depletion due to fluid losses such as vomiting or diarrhea or fluid shifts as in ascites; urinary tract obstruction due to tumor involvement; and/or intrinsic renal disease related to diabetes, hypertension, or chronic heart failure [89]. Exposure to concomitant nephrotoxic agents (e.g., acyclovir, allopurinol, aminoglycosides, amphotericin B, angiotensin-converting enzyme inhibitors, angiotensin receptor blockers, contrast agents, diuretics, foscarnet, ganciclovir, lithium, nonsteroidal anti-inflammatory drugs, proton pump inhibitors, sulfonamides, tacrolimus, and vancomycin) also increases the risk of acute kidney injury [88]. The following section discusses chemotherapy agents most commonly associated with kidney injury.

Chemotherapy Agents Implicated

Platinum-Based Compounds

Nephrotoxicity is a major dose-limiting side effect of cisplatin occurring in about 30% of patients. Nephrotoxicity results from tubular epithelial cell toxicity, vasoconstriction, and proinflammatory effects. These effects are due to a chloride ion in the cis position. Carboplatin and oxaliplatin do not contain this chloride and consequently are less nephrotoxic. The renal dysfunction manifests as hypomagnesemia, salt wasting, a Fanconi-like syndrome, and anemia which can progress and potentially be irreversible [89]. To minimize CIN, intravenous hydration, mannitol, and amifostine can be used [90, 91].

Alkylating Agents

Hemorrhagic cystitis (HC) can result from administration of cyclophosphamide or ifosfamide. The toxic metabolite, acrolein, is filtered by the kidneys and then concentrates in the bladder, inducing an inflammatory response. In addition, chloroacetaldehyde, a major metabolite of ifosfamide, is toxic to tubular cells [89]. Forced saline diuresis and mesna can be used to prevent HC in patients receiving highdose cyclophosphamide or ifosfamide. Patients should be instructed to drink fluids and void at first sensation as well as void at least once during the night [91]. Both agents can also cause syndrome of inappropriate antidiuretic hormone (SIADH) which results in hyponatremia. In addition, ifosfamide can cause Fanconi syndrome, characterized by a global transport defect in the proximal tubule of the kidney, and cause hypophosphatemia, polyuria, acidosis, hypokalemia, glycosuria, and proteinuria [90].

Antitumor Antibiotic

Renal toxicity with mitomycin can occur in up to 10% of patients and manifests as hemolytic uremic syndrome (HUS), resulting in thrombotic microangiopathy (TMA) from direct endothelial injury. The renal failure develops slowly and is usually associated with hypertension [89, 92].

Antimetabolites

High-dose methotrexate causes AKI in about 2% of patients secondary to ATN resulting from crystallization of the par-

ent drug and metabolites within the kidney. The AKI is usually reversible and can be minimized by ensuring appropriate hydration and high urine flow with urinary alkalinization (pH >7.0) [89]. Leucovorin is often given with high-dose methotrexate to reduce toxicities [90]. Pemetrexed has been associated with AKI, ATN, renal tubular acidosis, and diabetes insipidus. AKI is usually reversible upon drug discontinuation, but irreversible AKI and interstitial fibrosis have been reported [89, 93]. Gemcitabine may induce TMA through direct endothelial injury. It can also be caused by an immune-mediated mechanism, and if this is the case, the drug must be avoided for life [89, 93].

Targeted Therapy

Magnesium wasting is a major adverse effect of cetuximab therapy. Cetuximab is an antagonist of the epidermal growth factor receptor (EGFR). EGFR causes downregulation of the transient receptor potential cation channel, subfamily, member 6 (TRPM6), which results in magnesium wasting [94]. The hypomagnesemia usually resolves after drug discontinuation. To prevent adverse effects of hypomagnesemia, oral and/or intravenous replacement is necessary [89, 95].

Electrolyte Disorders

Electrolyte derangements are prevalent in cancer patients and can occur for a multitude of reasons. A common cause of fluid and electrolyte abnormalities in cancer patients is dehydration due to nausea, vomiting, diarrhea, or decreased appetite. Other reasons include tumor lysis syndrome, medication adverse effects, acute kidney injury, or complications of the cancer itself. Common electrolyte abnormalities in cancer patients are hyponatremia, hypomagnesemia, and hypo- or hyperkalemia [96]. Though many electrolyte abnormalities may not result from chemotherapeutic agents, many agents are well known to cause significant electrolyte disturbances, which will be the focus of this section. A detailed description of electrolyte disorders and chemotherapy agents implicated can be seen in Table 48.8 [97–100].

Pathophysiology

The mechanism by which chemotherapeutic agents cause electrolyte abnormality differs with each agent, ranging from renal wasting to direct tubular damage. Pathophysiologic details pertaining to each agent will be discussed within the sections below.

Chemotherapy Agents Implicated

Platinum-Based Chemotherapy

Platinum-based chemotherapeutic agents cause a wide variety of electrolyte abnormalities including hypomagnesemia, hypocalcemia, hypokalemia, and hyponatremia. Of all the platinum-based agents, cisplatin causes the most profound electrolyte disturbances, of which hypomagnesemia is most prevalent, occurring in up to 90% of patients [98]. The incidence appears to be dose-related and was found to reach 100% after the sixth cycle of chemotherapy in one study [101]. Hypomagnesemia occurs due to proximal tubular necrosis caused by cisplatin at the site of magnesium reabsorption, resulting in renal magnesium wasting [102]. Hypokalemia can also occur in the same manner [98].

Cisplatin-induced hypomagnesemia persists in many patients, up to years after cisplatin discontinuation [103]. Due to its high incidence of nephrotoxicity, cisplatin administration is usually preceded by judicious hydration with intravenous fluids. This induces an osmotic diuresis, which increases urinary magnesium excretion and can worsen hypomagnesemia [104]. Most patients are asymptomatic; however, muscle weakness and cramping, tetany, fatigue, seizures, arrhythmias, paralysis, and neurotoxicity can occur [98, 105]. Even if asymptomatic, it is important to correct because low serum magnesium can lead to refractory hypokalemia and hypocalcemia [98, 104, 106]. The persistence of hypomagnesemia has even been associated with decreased survival in ovarian cancer patients [95].

In addition to hypomagnesemia, hypokalemia, and hypocalcemia, platinum agents can also cause hyponatremia (in up to 43% treated with cisplatin) [98]. In a pharmacovigilance study where 19,901 adverse event reports were evaluated in non-small cell lung cancer patients, hyponatremia and hypokalemia were significantly associated with carboplatin exposure [107]. SIADH and renal salt wasting are likely the mechanisms by which platinum-based agents cause hyponatremia [98, 108]. Salt wasting can occur because of direct tubular damage from cisplatin, leading to a disruption in sodium and water reabsorption. This syndrome has been reported to occur in up to 10% of patients. One important consideration is to ensure an accurate diagnosis of hyponatremia, as treatment for SIADH and salt wasting syndrome differs significantly (free water restriction for the former and volume repletion for the latter) [108].

Cyclophosphamide

Hyponatremia resembling SIADH induced by cyclophosphamide was initially and is more commonly reported with high doses (30–50 mg/kg) [98]. However, this adverse effect has been reported with moderate doses and doses as low as 500 mg/m² and 10 mg/kg and even after a single dose of

Electrolyte disorder	Agents	Mechanism	Clinical presentation	Treatment ^a
Hyponatremia (serum sodium <135 mEq/L)	Carboplatin Cisplatin Ifosfamide Interferon Levamisole Melphalan Vinblastine Vincristine Vinorelbine Cyclophosphamide Methotrexate	Renal salt wasting or SIADH SIADH SIADH or water intoxication Alteration in body fluid volumes	Headache Nausea/vomiting Confusion Seizures Hypovolemic if renal salt wasting Euvolemic if SIADH	Renal salt wasting: Volume repletion with PO/IV sodium chloride supplementation SIADH: Fluid restriction Can consider low-dose diuretics Can consider oral sodium chloride If severe or acute symptoms, consider initial management with hypertonic, 3% sodium chloride infusion at a rate of 15–80 mL/h, or up to 1–2 mL/kg/hour with frequent sodium monitoring
Hypomagnesemia (serum magnesium <1.5 mg/dL)	Cetuximab Cisplatin Carboplatin Oxaliplatin Ifosfamide	Magnesium wasting (impaired reabsorption of magnesium in the kidneys)	Muscle cramps and weakness Paresthesias Tetany Tremulousness Arrhythmias Seizures	PO ^b /IV magnesium supplementation (e.g., magnesium oxide for PO, magnesium sulfate for IV) Mild to moderate hypomagnesium (1.0–1.5 mg/dL): 8–32 mEq Severe hypomagnesemia (<1.0 mg/dL): 32–64 mEq
Hypokalemia	Azacitidine Cisplatin Ifosfamide Streptozocin Abiraterone	Renal losses via renal tubular acidosis, Fanconi syndrome, or secondary to hypomagnesemia Excessive mineralocorticoid activity	Weakness Constipation EKG changes Arrhythmias	PO/IV potassium supplementation (e.g., potassium chloride) Mild to moderate hypokalemia (2.5–3.4 mEq/L): 20–40 mEq Severe hypokalemia (<2.5 mEq/L): 40–80 mEq Ensure adequate repletion of magnesium
Hypophosphatemia (serum phosphorus <2.7 mg/dL)	Azacitidine Ifosfamide Streptozocin	Proximal tubular damage impairing reabsorption of phosphorus, Fanconi syndrome	Respiratory distress Weakness Paresthesias Neurologic dysfunction Seizures	PO/IV phosphorus supplementation (i.e., sodium phosphate or potassium phosphate) Mild hypophosphatemia (2.3–2.7 mg/dL): 0.08–0.16 mmol/kg Moderate hypophosphatemia (1.5–2.2 mg/ dL): 0.16–0.32 mmol/kg Severe hypophosphatemia (<1.5 mg/dL): 0.32–0.64 mmol/kg
Hypocalcemia (corrected serum calcium ^c <8.6 mg/dL)	Cetuximab Cisplatin	Potentially secondary to hypomagnesemia (causing impaired parathyroid function) or via renal losses	Tetany Seizures	PO/IV calcium supplementation (e.g., calcium carbonate for PO, calcium gluconate for IV) Mild to moderate/asymptomatic: Calcium gluconate 1–2 g IV or PO supplementation Severe/symptomatic: Calcium chloride ^d 1 g IV or calcium gluconate 3 g IV Ensure adequate repletion of magnesium

Table 48.8	Electrolyte	disorders and	chemotherapy	agents	implicated	[97	-1	00	1
-------------------	-------------	---------------	--------------	--------	------------	-----	----	----	---

CrCl creatinine clearance, DMSO dimethyl sulfoxide, IV intravenous, mg milligrams, SC subcutaneous

^aTreatment recommendations are for patients with normal renal function; for patients with impaired renal function, administer 50% or less of the recommended dose

^bPO magnesium may cause diarrhea, especially in larger doses

^cCorrected serum calcium = serum calcium + $(0.8 \times [4 - \text{serum albumin}])$

^dRequires administration via central IV access because extravasation may cause tissue necrosis

^aSome literature suggests using cold compresses with taxanes

500 mg [83, 109]. Cyclophosphamide-associated SIADH is postulated to be caused by either an antidiuretic hormone (ADH)-like metabolite or a direct toxic effect on the renal distal tubule by cyclophosphamide or one of its metabolites [97, 110]. Hyponatremia presents variably depending

on the acuity and severity, causing symptoms ranging from headache, nausea, and vomiting to altered mental status, seizures, and coma [99]. SIADH in this setting generally occurs 4–12 h after drug administration and usually resolves within 24 h of drug discontinuation [97]. Treatment for SIADH centers on fluid restriction, but this may be challenging due to the recommendation for generous hydration to prevent cyclophosphamide-induced hemorrhagic cystitis [105].

Vinca Alkaloids

Of the vinca alkaloids, vincristine poses the highest risk of hyponatremia. This hyponatremia is thought to resemble SIADH and involves an inappropriate release of antidiuretic hormone in the setting of hypotonic hyponatremia [105]. The incidence is 1.3 in 100,000 treated patients and can be up to 44% of treated patients when given with azole antifungals due to a drug–drug interaction [98, 111]. The onset of hyponatremia occurs 1–2 weeks after administration and usually lasts for 2 weeks but can last up to 30 days. Hyponatremia induced by vinca alkaloids is usually reversible with proper treatment, but fatalities have been reported [97].

Ifosfamide

Fanconi syndrome is a well-known adverse effect of ifosfamide therapy as previously discussed [98, 112]. Fanconi syndrome involves damage to the proximal tubules resulting in impairment of electrolyte reabsorption, including phosphorus, bicarbonate, and glucose. The predominant concern is hypophosphatemia, which can be severe enough to cause rickets in the pediatric population [112]. Other signs and symptoms of hypophosphatemia include respiratory distress, neurologic dysfunction, and seizures [99]. Several risk factors for ifosfamide nephrotoxicity have been identified, including high cumulative doses of ifosfamide, concomitant cisplatin use, and nephrectomy [112]. Lastly, ifosfamide may also cause hypokalemia through renal wasting [98]. It is also important to note that manifestations of ifosfamide toxicities may not appear until years following therapy.

Cetuximab

Cetuximab was shown to cause hypomagnesemia in over 11% of patients in one retrospective study. Patients receiving a concomitant platinum-based agent had a more rapid and more significant decrease in magnesium levels at the end of 12 weeks [113]. The mechanism underlying cetuximabinduced hypomagnesemia appears to be impaired reabsorption of magnesium due to cetuximab's inhibition of epidermal growth factor receptor (EGFR), which aids in magnesium transport, thus resulting in magnesium wasting [114]. Although severe hypomagnesemia occurs in only 11% of patients, it has been shown that nearly half of patients who receive cetuximab develop some degree of hypomagnesemia [113]. Oftentimes, hypomagnesemia due to cetuximab use is refractory to oral magnesium replacement and requires intravenous replacement. Another study showed variable rates of hypomagnesemia based on cancertype with non-small cell lung cancer at the highest risk and colorectal cancer at the lowest risk [115]. Interestingly, hypomagnesemia was shown to be predictive for delayed disease progression and improved survival in colorectal cancer patients [114].

Treatment

It is important to be cognizant of the common electrolyte disorders that can occur with chemotherapy in order to monitor serum electrolytes accordingly. Treatment of chemotherapyinduced electrolyte disorders requires vigilant monitoring and judicious replacement or correction when warranted. Oftentimes, the culprit for the electrolyte disorder is difficult to pinpoint due to confounders such as concomitant medications and disease processes, GI losses from vomiting or diarrhea, dehydration, and poor nutrition status. When identified, treatment of these electrolyte disorders usually requires only replacement or correction of balance of water and electrolytes (e.g., fluid restriction for SIADH hyponatremia) but may require alterations in therapy such as chemotherapy dose reductions, transitioning to alternative agents within the same class, changing the administration schedule, or even discontinuing the agent. When replacing electrolytes such as magnesium, potassium, phosphorus, and calcium, the clinician should be mindful of the patient's renal function and adjust replacement accordingly for impaired function. Electrolytes should be replaced orally if the patient is able to tolerate oral intake and if the patient is asymptomatic or mildly symptomatic. For acute, severe, or symptomatic electrolyte disturbances, electrolyte replacement should be given intravenously. The reader is advised to refer to other references for a more detailed discussion of the treatment of electrolyte disorders [99, 116, 117].

Anaphylaxis

Patients with cancer are increasingly exposed to a wider range of chemotherapy and targeted agents. Increased exposure leads to a higher opportunity to develop severe hypersensitivity reactions such as anaphylaxis.

Pathophysiology

The exact mechanism by which hypersensitivity reactions occur is often unclear and may vary among agent [118]. Most reactions to chemotherapy and targeted agents are consistent with type I hypersensitivity based on the Gell and Coombs immunopathologic mechanism [118, 119] (see Table 48.9 [119, 120]).

	Type I	Type II	Type III	Type IV
Mediated	IgE	IgG or IgM Destruction of cells	Antigen-antibody complexes	T-cell (CD4 ⁺ or CD8 ⁺)
Timing	Seconds to minutes Can be delayed (1-72 h)	Days	Hours to days	Delayed (2-3 days)
Reactions	Anaphylaxis Laryngeal edema Bronchospasms Cutaneous reactions Nausea Vomiting	Hemolytic anemia Thrombocytopenia	Inflammation Serum sickness Vasculitis	Dermatologic
Causes	Hay fever Food IV contrast dye Latex Vaccines Insect bites/stings Medications such as chemotherapy and targeted agents	Transfusion reactions Methylene blue Heparin	Beta-lactams Quinidine Minocycline	Organ transplant rejection Poison ivy

 Table 48.9
 Types of hypersensitivity reactions [119, 120]

Clinical Features

Anaphylaxis is an allergic reaction characterized by multisystem involvement and is considered a type I mediated reaction that is seen with certain chemotherapy and targeted agents listed below. The initial symptoms of anaphylaxis are often nonspecific and include tachycardia, faintness, cutaneous flushing, urticaria, diffuse or localized pruritus, and a sensation of impending doom. These symptoms usually occur within minutes of administering the offending agent, but reactions may develop later. Biphasic or late-phase reactions that occur 1–72 h after the initial attack have been reported [121, 122].

Chemotherapy and Targeted Agents Implicated

Hypersensitivity reactions are more commonly associated with certain chemotherapeutic and targeted agents such as taxanes, platinum-containing compounds, epipodophyllotoxins, asparaginase, procarbazine, monoclonal antibodies, and, occasionally, doxorubicin and 6-mercaptopurine [120, 123]. Immediate, acute reactions from monoclonal antibodies have been reported in 5-10% of patients for rituximab, 2-3% for infliximab, and 0.6-5% for trastuzumab and have also been reported with omalizumab, natalizumab, basiliximab, abciximab, and cetuximab [123]. Hypersensitivity symptoms to taxanes usually develop in the first few minutes of the infusion and mostly occur on the first or second exposure. Platinum-containing compound reactions usually occur at the time of cancer reoccurrence or after the patient has been exposed for at least six cycles [123].

Treatment

Early and rapid assessment is crucial and every minute counts. If a patient is unresponsive and pulseless, cardiopulmonary resuscitation should begin, and the appropriate advanced cardiovascular life support (ACLS) algorithm should be used. Advanced airway management with oxygen should be established and maintained without any delay. Circulation should be supported initially with a rapid fluid challenge of 500–2000 mL of 0.9% sodium chloride [122, 124].

Epinephrine treatment is also used to help maintain circulation in patients experiencing anaphylaxis. In a Cochrane systematic review, no randomized controlled trials using epinephrine were identified. The widespread use of epinephrine in anaphylaxis is based on non-randomized studies and expert opinion and is recommended in all anaphylaxis guidelines published to date [122]. Other types of studies include fatality studies; most people who died from anaphylaxis did not receive an epinephrine injection before cardiac arrest [125].

The adult dose of epinephrine for weight-based dosing is 0.01 mg/kg, to a maximum dose of 0.5 mg intramuscularly (IM) repeated every 5–15 min, as needed for unresolved anaphylaxis symptoms [122]. Epinephrine IM injections into the lateral aspect of the thigh have been reported to provide more rapid absorption and higher plasma epinephrine levels than IM or SC administration in the arm [126]. When anaphylaxis is not responding to repeated epinephrine IM or SC doses, intravenous (IV) epinephrine should be considered [122, 127]. However, caution should be taken when using this route. In an observational study by Campbell and colleagues, they noted that all epinephrine overdoses occurred in patients receiving IV bolus, with more arrhythmia and ischemic events than with IM and SC injections [128].

Adjunctive therapy includes H_1 -antihistamines, H_2 antihistamines, and glucocorticoids. None of these therapeutic modalities have rigorous evidence to support their use, and they are second-line or adjuvant therapies in most guidelines. The first-generation, H₁-antihistamine, diphenhydramine, has been used because of its availability in IV formulation and its efficacy in relieving urticaria and itching within minutes after oral or IV administration. Due to its ability to cross the blood-brain barrier readily, diphenhydramine causes drowsiness and impaired cognitive function. Diphenhydramine given orally, IM, or slow IV in a dose of 25–50 mg has been suggested [129]. There are no studies supporting administration of H2-antihistamines or recommending one over another; however, this class of drugs has a low potential to harm and may be helpful in the management of anaphylaxis. Giving an H₂- with a H₁-antihistamine may decrease urticaria, flushing, headache, hypotension, and rhinorrhea [125].

Glucocorticoids are traditionally administered to help reduce symptoms and to prevent biphasic anaphylactic reactions that may occur. The onset of the glucocorticoid action can be anywhere from 4 to 6 h. They are not the drug of choice for the initial phase of anaphylaxis and due to low evidence are not recommended for use in the 2020 anaphylaxis practice parameter publication [122]. Figure 48.1 is an algorithm detailing steps for treating patients with signs and symptoms of anaphylaxis.

Early recognition and treatment of anaphylaxis saves lives. There is little to no supporting evidence for the treatment of anaphylaxis, but epinephrine has shown to be beneficial, and death has been associated with failure to use epinephrine before cardiac arrest.

Extravasation

Extravasation is a well-recognized complication of intravenous chemotherapy [130]. The incidence of accidental extravasation of intravenous drugs into the tissue can be anywhere from 0.1% to 6.5% [130, 131].

Clinical Features

Extravasation is the unintentional instillation, leakage, passage, or escape of fluid or drug from a blood vessel into surrounding tissue. This may result in varying degrees pain, necrosis, and tissue sloughing. The degree of tissue damage is related to the properties of the drug extravasated, the duration of tissue exposure, and the amount of drug that infiltrated [132]. Chemotherapy drugs are classified into three categories according to their potential cause of tissue



Fig. 48.1 Algorithm detailing steps for treating patients with signs and symptoms of anaphylaxis

damage: vesicant, irritant, and non-vesicant. A vesicant is any agent with potential to cause tissue destruction, blistering, severe injury, or tissue necrosis, when extravasated [132, 133]. An irritant is any agent that can cause inflammation or irritation characterized by aching, tightness, and phlebitis but without necrosis. Non-vesicants are drugs that rarely produce acute reactions or destroy tissue when they infiltrate.

Chemotherapy Agents Implicated

Chemotherapy agents are often classified into the above three categories (Table 48.10 [130, 131, 133–135]). Every attempt

 Table 48.10
 Classification of chemotherapy agents according to their vascular damage potential [130, 131, 133–135]

Vesicants	Irritants	Non-vesicants
Cisplatin ^a (if >20 mL	Arsenic trioxide ^b	Arsenic trioxide ^b
of 0.5 mg/mL)	Bleomycin ^b	Asparaginase
Docetaxel ^a	Bortezomib ^b	Bleomycin ^b
Dactinomycin	Busulfan	Bortezomib ^b
Daunorubicin	Carmustine	Cladribine ^b
Doxorubicin	Carboplatin	Cyclophosphamide
Epirubicin	Cladribine ^b	Cytarabine
Idarubicin	Cisplatin ^a (if	Fludarabine
Mechlorethamine	concentration less	Gemcitabine ^b
Mitomycin C	than 0.5 mg/mL)	Ifosfamide ^b
Mitoxantrone	Dacarbazine	Interferons
Oxaliplatin ^a	Docetaxel ^a	Interleukin-2
Paclitaxel ^a	Etoposide	Melphalan ^b
Vinblastine	Fluorouracil	Methotrexate
Vincristine	Gemcitabine ^b	Monoclonal
Vindesine	Ifosfamide ^b	antibodies
Vinorelbine	Irinotecan	Pemetrexed
	Ixabepilone	Pentostatin
	Liposomal cytarabine	Raltitrexed
	Liposomal	Temsirolimus
	daunorubicin	Thiotepa ^b
	Liposomal	
	doxorubicin	
	Liposomal vincristine	
	Melphalan ^b	
	Mitoxantrone ^a	
	Oxaliplatin ^a	
	Paclitaxel ^a	
	Paclitaxel,	
	nanoparticle albumin	
	bound	
	Plicamycin	
	Streptozocin ^a	
	Teniposide	
	Thiotepa ^b	
	Topotecan	

^aHas been described as both an irritant and vesicant in the literature ^bHas been described as both irritant and non-vesicant in the literature

should be made to minimize the risk of extravasation. This is a collaborative practice involving physicians, nurses, pharmacists, patients, and patients' caregivers. All healthcare professionals should adhere to institutional policies and procedures to prevent extravasation. Unfortunately, not all extravasations are preventable; however, individualized risk factors for extravasation are recognized (Table 48.11 [134, 136, 137]). All parties involved should be cognizant of the signs and symptoms of chemotherapy extravasation. These may include stinging or burning pain, as well as erythema, induration, blister, or vesicle formation. Around the injection site, there may be venous discoloration, swelling, or leakage of fluid. The flow rate of the infusion may change with increased resistance that cannot be explained. Blood return from the infusion line may be impaired or impossible [132, 134]. Chemotherapy extravasation reactions may not manifest until hours, days, or even months after the infusion has been stopped or completed [138].

 Table 48.11
 Patient risk factors for extravasation [134, 136, 137]

Patient associated	Others
Small and fragile veins	Insufficient training of staff,
Vascular disease, Raynaud's disease;	poor technique
peripheral neuropathies; peripheral	Butterfly, metal, or
vascular disease such as diabetes	large-gauged needles
mpaired lymph flow and venous	Inadequate secured needle
circulation	Veins used adjacent to
Superior vena cava syndrome	tendons, nerves, or arteries
Locally infiltrating tumors	Needle location (e.g.,
Age—elderly and young at the highest	antecubital fossa, wrist or
isk	dorsum of hand)
Restlessness, agitation, altered mental	Inappropriate needle length
status, or confusion	Catheter failure
Cerebral vascular accident	Multiple attempts at
Coagulation abnormalities	cannulation
Dbesity	Irritant and vesicant drugs
Prior extravasation injury	Prolonged infusions
	Infiltration volume
	Multiple treatments of
	chemotherapy agents
	Previous vinca alkaloids
	administration
	Radiation therapy-current
	or past

Treatment: Pharmacologic and Non-pharmacologic

Extravasation of chemotherapy requires prompt management. Delays in vesicant extravasation treatment can cause further tissue damage, necrosis, and pain that require surgical intervention. The type of chemotherapy, drug concentration, amount of drug extravasated, pH, and osmolarity play a role in severity of tissue injury [132]. The general approach to extravasation management is early recognition and treatment to minimize the extent of extravasation and prevent further tissue damage. For patients reporting burning, pain, or any other infusion-related issue, such as swelling, increased resistance, or leakage of fluid around the site, the chemotherapy infusion stopped immediately. The infusion site should be inspected for extravasation. Patients with extravasation due to a vesicant and any large-volume extravasation or those having worsening pain at extravasation site or evidence of ulceration or tissue necrosis should present to the ED for evaluation and consideration of plastic surgery consultation.

Institutions administering chemotherapy infusions should have policies and procedures in place to ensure early intervention for extravasations. An extravasation kit should be available in the chemotherapy infusion areas. These contain both non-pharmacologic and pharmacologic treatment supplies, including disposable syringes, small-gauged needles, cold-hot packs, gauze pads, adhesive plaster, sterile and protective gloves, and medications (e.g., dexrazoxane, dimethyl sulfoxide [DMSO] 50–99%, and hyaluronidase). Pre-prepared kits allow providers to initiate treatment in the infusion areas and minimize any delay in care.

Fig. 48.2 Treatment steps that should be completed when patients present with signs and symptoms of chemotherapy extravasation [130, 134]	Step 1	Stop chemotherapy infusion immediately. Leave needle in place. Identify chemotherapy agent.
	Step 2	Aspirate any additional drug out of catheter with syringe. Record volume aspirated.
	Step 3	Remove needle.
	Step 4	Mark the area and take photo of extravasation; notify physician.
	Step 5	Administer non-pharmacologic and pharmacologic treatment based on chemotherapy (see Table 48.13).
	Step 6	Elevate limb; administer additional medications as needed for pain. Assess area for necrotic tissue; consider surgical management.

Figure 48.2 [130, 134] denotes all treatment steps that should be completed when patients present with signs and symptoms of chemotherapy extravasation. The infusion should be discontinued as soon as extravasation is suspected. The infusion line should be disconnected, and the needle should be left in place to allow for aspiration of residual chemotherapy. Extensive or large-volume extravasation of irritants and non-vesicants should be evaluated by a plastic surgeon for further intervention and management. Small extravasation of irritants and non-vesicants should be monitored over several days to ensure that the skin is viable and pain resolves. A plastic surgery consult is recommended if the extravasation site develops full-thickness necrosis and chronic nonhealing ulceration or if the patient experiences persistent or worsening pain [139]. Vesicants should be treated with both non-pharmacologic and pharmacologic approaches. Physicians should be notified immediately with information regarding the chemotherapy agent to ensure appropriate and prompt treatment initiation. The extravasation area should be marked with a pen to monitor for extravasation spread. If the first four steps have not been completed prior to ED presentation, it is crucial that they be completed as soon as possible.

Various non-pharmacologic and pharmacologic treatments have been utilized in the past to help minimize the damage of chemotherapy agent extravasations. Available studies report varying degrees of success in managing extravasations. It should be noted that only a handful of treatment options or "antidotes" exist that can be injected or topically applied to the affected area. Vesicant chemotherapy agents can be divided into two categories: DNA binding versus non-DNA binding (Table 48.12), and treatment options for these categories differ [130].

DNA binding vesicants, such as anthracyclines, bind to the DNA in the cells of healthy tissue and promote cell death. DNA-chemotherapy complexes are released from the dead cells in the tissue and are taken up by adjacent healthy cells via endocytosis. This process is cyclic and can continue for

 Table 48.12
 Classification of vesicant chemotherapy agents [130]

Classification	Examples	Pharmacologic		
Classification	Examples	ueaunent		
DNA binding				
Alkylating	Mechlorethamine	Sodium thiosulfate		
agents				
Anthracyclines	Daunorubicin, doxorubicin,	DMSO or		
	epirubicin, idarubicin	dexrazoxane		
Others	Mitoxantrone, mitomycin	DMSO		
	dactinomycin			
Non-DNA bindin				
Vinca alkaloids	Vinblastine, vincristine,	Hyaluronidase		
	vindesine, vinorelbine			
Taxanes	Docetaxel, paclitaxel	Hyaluronidase		

weeks to months after the incident [130]. If left untreated, the cell damage spreads larger and deeper, becoming more painful with time. Debridement may be required if severe injury results in tissue necrosis. Localizing the offending agent can be achieved by cooling the area with a dry cool compress or cold gel pack. This procedure may lessen the pain by constricting vessels and preventing the offending agent from spreading to surrounding healthy tissue. There is insufficient published evidence to support the efficacy of this treatment, but it may be beneficial in reducing discomfort, burning sensation, and tenderness [136]. The next step is to neutralize the offending agent, and this will depend on the extravasated chemotherapy agent (Table 48.13 for dosing) [130, 140–144].

Non-binding DNA vesicants, such as vinca alkaloids, have an indirect effect on healthy tissue. Non-DNA-binding chemotherapy agents are metabolized in the tissue and are more easily neutralized [130]. The injury caused by these agents is localized, is mildly to moderately painful, and improves over time [145]. Local warming is preferred with this group of agents to increase blood flow to the area. Warm compresses distribute the chemotherapy agent and promotes its absorption [131]. However, some controversies have arisen regarding this technique. Some literature reports wors-
Non-pharmacologic treatment	Pharmacologic treatment dosing		
Dry cold compress (3-day course)	Dexrazoxane (3-day course)		
Immediately for 20 min and four times daily—do not apply 15 min prior to or during dexrazoxane infusion	Within 6 h: 1000 mg/m ² (max 2000 mg) IV Day 2: 1000 mg/m ² (max 2000 mg) IV Day 3: 500 mg/m ² (max 1000 mg) IV Maximum dose of 2000 mg per day. Doses are 24 h apart and infused over 1–2 h in opposite arm If CrCl is less than 40 mL/min—decrease dose by 50% or DMSO 50–99%		
	Immediately apply topically 1–2 mL with cotton swab over area twice the size of that affected and allow to air dry every 6–8 h for 7–14 days; do not apply if using dexrazoxane		
Dry cold compress	Sodium thiosulfate 1/6 M		
	1/6 M = 4 mL 10% sodium thiosulfate + 6 mL water		
	Inject 2 mL for each 100 mg of extravasated cisplatin through existing needle; if needle has been removed, inject 1 mL SC (0.1 mL doses clockwise around extravasation using small-gauged [25 or less] needle); may repeat SC dose several times over the next 3–4 h		
Dry cold compress	Sodium thiosulfate 1/6 M		
	1/6 M = 4 mL 10% sodium thiosulfate + 6 mL water		
	Inject 2 mL for each 1 mL of extravasated mechlorethamine into existing needle; if needle has been removed, inject 1 mL SC (0.1 mL doses clockwise around extravasation using small-gauged [25 or less] needle); may repeat SC dose several times over the next 3–4 h		
Dry cold compress (3-day course)	Topical DMSO 50–99%		
Immediately for 20 min and four times daily	Immediately apply topically 1–2 mL with cotton swab over area twice the size of that affected and allow to air dry every 6–8 h for 7–14 days		
Dry warm ^a compress (3-day course)	Hyaluronidase		
Immediately for 20 min and four times daily	Inject 1–6 mL of 150 units/mL solution through the existing needle; if needle has been removed, inject 1 mL of solution per 1 mL of extravasated chemotherapy agent SC in a clockwise manner into multiple sites of the extravasation area using small-gauged (25 or less) needle; may repeat SC dose several times over the next 3–4 h		
	Non-pharmacologic treatment Dry cold compress (3-day course) Immediately for 20 min and four times daily—do not apply 15 min prior to or during dexrazoxane infusion Dry cold compress Dry cold compress Dry cold compress Dry cold compress (3-day course) Immediately for 20 min and four times daily Dry warm ^a compress (3-day course) Immediately for 20 min and four times daily		

Table 48.13 Treatment after extravasation [130, 140–144]

CrCl creatinine clearance, *DMSO* dimethyl sulfoxide, *IV* intravenous, *mg* milligrams, *SC* subcutaneous ^aSome literature suggests using cold compresses with taxanes

ening injury at the extravasation site with warm compresses and rather recommends cold therapy to limit the diffusion of chemotherapy [135, 146, 147]. Pharmacologic treatment used for these agents helps dilute the offending agent (see Table 48.13 for dosing) [130, 140–144].

DNA-Binding Agents

Dexrazoxane

Dexrazoxane is approved for anthracycline extravasations. Dexrazoxane inhibits DNA topoisomerase II—the target of anthracyclines—minimizing pain and oxidative damage in the tissue by chelating metal ions from anthracycline [148]. Dexrazoxane is well tolerated and highly effective in avoiding surgical resection after anthracycline extravasation [140, 141]. Patients receiving dexrazoxane may experience bone marrow suppression (underlying disease and chemotherapy), mild transient elevation of liver enzymes (occurs in ~25% of patients), nausea (20%), and local infusion site pain [140]. Concomitant use with dimethyl sulfoxide may reduce the efficacy of dexrazoxane [149].

DMSO

Dimethyl sulfoxide (DMSO) is a topically applied solvent that increases skin permeability, promotes absorption of extravasated vesicants, and scavenges free radicals. DMSO has been studied in various amounts and concentrations (50–100%), application frequencies (every 2–8 h), durations of treatment (2–14 days), and for several different types of chemotherapy extravasations [137]. DMSO should be applied immediately and reapplied every 6–8 h for 7–14 days to decrease risk of ulceration and need for surgical debridement. Pain and erythema usually resolved in the first 1–2 weeks, and DMSO may be used for up to 6 weeks. Reported side effects include mild pigmentation in the area, mild discomfort at the injection site, and a characteristic garlic breath odor [142]. Local DMSO in combination with dexrazoxane should be avoided. DMSO in combination may lessen the effects of dexrazoxane [135, 149]. Toxicity associated with mitomycin C may also be prevented by DMSO used topically [143, 150].

Sodium Thiosulfate

Mechlorethamine can cause severe ulceration and tissue damage. Sodium thiosulfate 1/6 M (0.17 M) solution is the only antidote currently available and recommended for extravasation of mechlorethamine or concentrated cisplatin (>20 mL of greater than 0.5 mg/mL) (see Table 48.13). When given subcutaneously into the extravasation area, sodium thiosulfate binds and neutralizes the extravasated vesicant and reduces production of hydroxyl radicals [151].

Non-DNA-Binding Agents

Hyaluronidase

Hyaluronidase can be used to treat vinca alkaloids or taxane extravasation [135, 144, 152]. Hyaluronidase is a protein enzyme that helps degrade hyaluronic acid that holds tissue planes together, rapidly dilutes the offending agent, and enhances drug absorption. Hyaluronidase doses range from 150 to 200 units, administered into the indwelling catheter or subcutaneously around the extravasation site [144, 149]. It has been shown to decrease pain and the amount of tissue injury over several days [144].

Surgical Intervention

The optimal timing of surgical intervention is unknown. Only one-third of vesicant extravasation in the extremities actually results in ulceration. Therefore, routine surgical intervention should not be the initial treatment [153]. Ulceration can progress to necrosis and infection. A plastic surgical consultation is recommended if the patient experiences a large-volume vesicant extravasation (not defined) and severe pain, if healing has not occurred 1–3 weeks after extravasation, or if there is early necrosis present [132, 135]. Surgical interventions may include debridement, fasciotomy, reconstruction, and grafting.

Saline flushing or washout/flush-out techniques have been described in the literature. There is some evidence that this may be beneficial. Several techniques have been used alone and in combination with other treatments [136]. The flush-out technique is usually performed by plastic surgeons, and the most recent described technique involves making several small-stab incisions and administering large volumes of 0.9% sodium chloride to dilute and flush out the extravasated drug [154, 155]. Saline washouts completed within 24 h of extravasation have been shown more effective than those completed more than 24 h post extravasation.

References

- Stone JB, DeAngelis LM. Cancer-treatment-induced neurotoxicity– focus on newer treatments. Nat Rev Clin Oncol. 2016;13(2):92–105.
- Verstappen CC, Heimans JJ, Hoekman K, Postma TJ. Neurotoxic complications of chemotherapy in patients with cancer: clinical signs and optimal management. Drugs. 2003;63(15):1549–63.
- 3. Chamberlain MC. Neurotoxicity of cancer treatment. Curr Oncol Rep. 2010;12(1):60–7.
- 4. Howard SC, McCormick J, Pui CH, Buddington RK, Harvey RD. Preventing and managing toxicities of high-dose methotrexate. Oncologist. 2016;21(12):1471–82.
- Bhojwani D, Sabin ND, Pei D, Yang JJ, Khan RB, Panetta JC, et al. Methotrexate-induced neurotoxicity and leukoencephalopathy in childhood acute lymphoblastic leukemia. J Clin Oncol. 2014;32(9):949–59.
- Quinn CT, Griener JC, Bottiglieri T, Hyland K, Farrow A, Kamen BA. Elevation of homocysteine and excitatory amino acid neurotransmitters in the CSF of children who receive methotrexate for the treatment of cancer. J Clin Oncol. 1997;15(8):2800–6.
- 7. Rinne ML, Lee EQ, Wen PY. Central nervous system complications of cancer therapy. J Support Oncol. 2012;10(4):133–41.
- Magge RS, DeAngelis LM. The double-edged sword: neurotoxicity of chemotherapy. Blood Rev. 2015;29(2):93–100.
- Omuro AMP, Ben-Porat LS, Panageas KS, Kim AK, Correa DD, Yahalom J, et al. Delayed neurotoxicity in primary central nervous system lymphoma. Arch Neurol. 2005;62(10):1595–600.
- Hasle H. Cerebellar toxicity during cytarabine therapy associated with renal insufficiency. Cancer Chemother Pharmacol. 1990;27(1):76–8.
- Tran PN, Kong XT. Cytarabine induced acute cerebellar syndrome during hyper-cvad treatment for Bb-cell acute lymphoblastic leukemia. Case Rep Neurol. 2017;9(1):114–20.
- Ajithkumar T, Parkinson C, Shamshad F, Murray P. Ifosfamide encephalopathy. Clin Oncol (R Coll Radiol). 2007;19(2):108–14.
- Pelgrims J, De Vos F, Van den Brande J, Schrijvers D, Prové A, Vermorken JB. Methylene blue in the treatment and prevention of ifosfamide-induced encephalopathy: report of 12 cases and a review of the literature. Br J Cancer. 2000;82(2):291–4.
- Sood C, O'Brien PJ. 2-Chloroacetaldehyde-induced cerebral glutathione depletion and neurotoxicity. Br J Cancer Suppl. 1996;27:S287–93.
- Patel PN. Methylene blue for management of Ifosfamide-induced encephalopathy. Ann Pharmacother. 2006;40(2):299–303.
- Eberly AL, Anderson GD, Bubalo JS, McCune JS. Optimal prevention of seizures induced by high-dose busulfan. Pharmacotherapy. 2008;28(12):1502–10.
- Vassal G, Deroussent A, Hartmann O, Challine D, Benhamou E, Valteau-Couanet D, et al. Dose-dependent neurotoxicity of highdose busulfan in children: a clinical and pharmacological study. Cancer Res. 1990;50(19):6203–7.
- Ciurea SO, Andersson BS. Busulfan in hematopoietic stem cell transplantation. Biol Blood Marrow Transplant. 2009;15(5): 523–36.
- Brydøy M, Oldenburg J, Klepp O, Bremnes RM, Wist EA, Wentzel-Larsen T, et al. Observational study of prevalence of long-term Raynaud-like phenomena and neurological side effects in testicular cancer survivors. J Natl Cancer Inst. 2009;101(24):1682–95.

- Avan A, Postma TJ, Ceresa C, Avan A, Cavaletti G, Giovannetti E, et al. Platinum-induced neurotoxicity and preventive strategies: past, present, and future. Oncologist. 2015;20(4):411–32.
- Rademaker-Lakhai JM, Crul M, Zuur L, Baas P, Beijnen JH, Simis YJ, et al. Relationship between cisplatin administration and the development of ototoxicity. J Clin Oncol. 2006;24(6):918–24.
- 22. Fung C, Dinh P Jr, Ardeshir-Rouhani-Fard S, Schaffer K, Fossa SD, Travis LB. Toxicities associated with cisplatin-based chemotherapy and radiotherapy in long-term testicular cancer survivors. Adv Urol. 2018;2018:8671832.
- Li SH, Chen WH, Tang Y, Rau KM, Chen YY, Huang TL, et al. Incidence of ischemic stroke post-chemotherapy: a retrospective review of 10,963 patients. Clin Neurol Neurosurg. 2006;108(2):150–6.
- Singh G, Rees JH, Sander JW. Seizures and epilepsy in oncological practice: causes, course, mechanisms and treatment. J Neurol Neurosurg Psychiatry. 2007;78(4):342–9.
- Brewer JR, Morrison G, Dolan ME, Fleming GF. Chemotherapyinduced peripheral neuropathy: current status and progress. Gynecol Oncol. 2016;140(1):176–83.
- 26. Pike CT, Birnbaum HG, Muehlenbein CE, Pohl GM, Natale RB. Healthcare costs and workloss burden of patients with chemotherapy-associated peripheral neuropathy in breast, ovarian, head and neck, and nonsmall cell lung cancer. Chemother Res Pract. 2012;2012:913848.
- Zajączkowska R, Kocot-Kępska M, Leppert W, Wrzosek A, Mika J, Wordliczek J. Mechanisms of chemotherapy-induced peripheral neuropathy. Int J Mol Sci. 2019;20(6):1451.
- Seretny M, Currie GL, Sena ES, Ramnarine S, Grant R, MacLeod MR, et al. Incidence, prevalence, and predictors of chemotherapyinduced peripheral neuropathy: a systematic review and metaanalysis. Pain. 2014;155(12):2461–70.
- 29. Molassiotis A, Cheng HL, Lopez V, Au JSK, Chan A, Bandla A, et al. Are we mis-estimating chemotherapy-induced peripheral neuropathy? Analysis of assessment methodologies from a prospective, multinational, longitudinal cohort study of patients receiving neurotoxic chemotherapy. BMC Cancer. 2019;19(1):132.
- Staff NP, Grisold A, Grisold W, Windebank AJ. Chemotherapyinduced peripheral neuropathy: a current review. Ann Neurol. 2017;81(6):772–81.
- Durand JP, Deplanque G, Montheil V, Gornet JM, Scotte F, Mir O, et al. Efficacy of venlafaxine for the prevention and relief of oxaliplatin-induced acute neurotoxicity: results of EFFOX, a randomized, double-blind, placebo-controlled phase III trial. Ann Oncol. 2012;23(1):200–5.
- Clinical pharmacology [database online]. Tampa: Elsevier; 2020; http://www.clinicalpharmacology.com. (Accessed 12 Jul 2020). Subscription required to view.
- Gewandter JS, Brell J, Cavaletti G, Dougherty PM, Evans S, Howie L, et al. Trial designs for chemotherapy-induced peripheral neuropathy prevention: ACTTION recommendations. Neurology. 2018;91(9):403–13.
- 34. Yang YH, Lin JK, Chen WS, Lin TC, Yang SH, Jiang JK, et al. Duloxetine improves oxaliplatin-induced neuropathy in patients with colorectal cancer: an open-label pilot study. Support Care Cancer. 2012;20(7):1491–7.
- Matsuoka H, Makimura C, Koyama A, Otsuka M, Okamoto W, Fujisaka Y, et al. Pilot study of duloxetine for cancer patients with neuropathic pain non-responsive to pregabalin. Anticancer Res. 2012;32(5):1805–9.
- 36. Smith EM, Pang H, Cirrincione C, Fleishman S, Paskett ED, Ahles T, et al. Effect of duloxetine on pain, function, and quality of life among patients with chemotherapy-induced painful peripheral neuropathy: a randomized clinical trial. JAMA. 2013;309(13):1359–67.
- Battaglini E, Park SB, Barnes EH, Goldstein D. A double blind, placebo controlled, phase II randomised cross-over trial investigat-

ing the use of duloxetine for the treatment of chemotherapy-induced peripheral neuropathy. Contemp Clin Trials. 2018;70:135–8.

- Barton DL, Wos EJ, Qin R, Mattar BI, Green NB, Lanier KS, et al. A double-blind, placebo-controlled trial of a topical treatment for chemotherapy-induced peripheral neuropathy: NCCTG trial N06CA. Support Care Cancer. 2011;19(6):833–41.
- 39. Rao RD, Michalak JC, Sloan JA, Loprinzi CL, Soori GS, Nikcevich DA, et al. Efficacy of gabapentin in the management of chemo-therapy-induced peripheral neuropathy: a phase 3 randomized, double-blind, placebo-controlled, crossover trial (N00C3). Cancer. 2007;110(9):2110–8.
- Saif MW, Syrigos K, Kaley K, Isufi I. Role of pregabalin in treatment of oxaliplatin-induced sensory neuropathy. Anticancer Res. 2010;30(7):2927–33.
- 41. Gradishar W, Vokes E, Schilsky R, Weichselbaum R, Panje W. Vascular events in patients receiving high-dose infusional 5-fluorouracil-based chemotherapy: the University of Chicago experience. Med Pediatr Oncol. 1991;19(1):8–15.
- Li K, Giustini D, Seely D. A systematic review of acupuncture for chemotherapy-induced peripheral neuropathy. Curr Oncol. 2019;26(2):e147–e54.
- 43. McCrary JM, Goldstein D, Sandler CX, Barry BK, Marthick M, Timmins HC, et al. Exercise-based rehabilitation for cancer survivors with chemotherapy-induced peripheral neuropathy. Support Care Cancer. 2019;27(10):3849–57.
- 44. James PA, Oparil S, Carter BL, Cushman WC, Dennison-Himmelfarb C, Handler J, et al. 2014 evidence-based guideline for the management of high blood pressure in adults: report from the panel members appointed to the Eighth Joint National Committee (JNC 8). JAMA. 2014;311(5):507–20.
- Mouhayar E, Salahudeen A. Hypertension in cancer patients. Tex Heart Inst J. 2011;38(3):263–5.
- Małyszko J, Małyszko M, Kozlowski L, Kozlowska K, Małyszko J. Hypertension in malignancy-an underappreciated problem. Oncotarget. 2018;9(29):20855–71.
- Fraeman KH, Nordstrom BL, Luo W, Landis SH, Shantakumar S. Incidence of new-onset hypertension in cancer patients: a retrospective cohort study. Int J Hypertens. 2013;2013:379252.
- Peixoto AJ. Acute severe hypertension. N Engl J Med. 2019;381(19):1843–52.
- 49. Whelton PK, Carey RM, Aronow WS, Casey DE Jr, Collins KJ, Dennison Himmelfarb C, et al. 2017 ACC/AHA/AAPA/ABC/ ACPM/AGS/APhA/ASH/ASPC/NMA/PCNA guideline for the prevention, detection, evaluation, and management of high blood pressure in adults: a report of the American College of Cardiology/ American Heart Association Task Force on Clinical Practice Guidelines. Hypertension. 2018;71(6):e13–e115.
- Arakawa-Todo M, Yoshizawa T, Zennami K, Nishikawa G, Kato Y, Kobayashi I, et al. Management of adverse events in patients with metastatic renal cell carcinoma treated with sunitinib and clinical outcomes. Anticancer Res. 2013;33(11):5043–50.
- Tocci G, Presta V, Volpe M. Hypertensive crisis management in the emergency room: time to change? J Hypertens. 2020;38(1): 33–4.
- 52. Powers WJ, Rabinstein AA, Ackerson T, Adeoye OM, Bambakidis NC, Becker K, et al. Guidelines for the early management of patients with acute ischemic stroke: 2019 update to the 2018 guidelines for the early management of acute ischemic stroke: a guideline for healthcare professionals from the American Heart Association/ American Stroke Association. Stroke. 2019;50(12):e344–418.
- Chobanian AV, Bakris GL, Black HR, Cushman WC, Green LA, Izzo JL Jr, et al. Seventh report of the joint national committee on prevention, detection, evaluation, and treatment of high blood pressure. Hypertension. 2003;42(6):1206–52.
- Chaikof EL, Dalman RL, Eskandari MK, Jackson BM, Lee WA, Mansour MA, et al. The society for vascular surgery practice guide-

lines on the care of patients with an abdominal aortic aneurysm. J Vasc Surg. 2018;67(1):2–77, e2.

- 55. Hemphill JC 3rd, Greenberg SM, Anderson CS, Becker K, Bendok BR, Cushman M, et al. Guidelines for the management of spontaneous intracerebral hemorrhage: a guideline for healthcare professionals from the American Heart Association/American Stroke Association. Stroke. 2015;46(7):2032–60.
- 56. Braverman AC. Acute aortic dissection: clinician update. Circulation. 2010;122(2):184–8.
- 57. Hiratzka LF, Bakris GL, Beckman JA, Bersin RM, Carr VF, Casey DE, et al. 2010 ACCF/AHA/AATS/ACR/ASA/SCA/SCAI/SIR/STS/SVM guidelines for the diagnosis and management of patients with thoracic aortic disease. A report of the American College of Cardiology Foundation/American Heart Association Task Force on Practice Guidelines, American Association for Thoracic Surgery, American College of Radiology, American Stroke Association, Society of Cardiovascular Anesthesiologists, Society for Cardiovascular Angiography and Interventions, Society of Interventional Radiology, Society of Thoracic Surgeons, and Society for Vascular Medicine. J Am Coll Cardiol. 2010;55(14):e27–e129.
- 58. Yancy CW, Jessup M, Bozkurt B, Butler J, Casey DE Jr, Drazner MH, et al. 2013 ACCF/AHA guideline for the management of heart failure: a report of the American College of Cardiology Foundation/ American Heart Association Task Force on Practice Guidelines. J Am Coll Cardiol. 2013;62(16):e147–239.
- Rhoney D, Peacock WF. Intravenous therapy for hypertensive emergencies, part 2. Am J Health Syst Pharm. 2009;66(16):1448–57.
- Vaughan CJ, Delanty N. Hypertensive emergencies. Lancet. 2000;356(9227):411–7.
- Peacock WF, Hilleman DE, Levy PD, Rhoney DH, Varon J. A systematic review of nicardipine vs labetalol for the management of hypertensive crises. Am J Emerg Med. 2012;30(6):981–93.
- Hecht JP, Richards PG. Continuous-infusion labetalol vs nicardipine for hypertension management in stroke patients. J Stroke Cerebrovasc Dis. 2018;27(2):460–5.
- Davis SM, Broderick J, Hennerici M, Brun NC, Diringer MN, Mayer SA, et al. Hematoma growth is a determinant of mortality and poor outcome after intracerebral hemorrhage. Neurology. 2006;66(8):1175–81.
- 64. Anderson CS, Huang Y, Wang JG, Arima H, Neal B, Peng B, et al. Intensive blood pressure reduction in acute cerebral haemorrhage trial (INTERACT): a randomised pilot trial. Lancet Neurol. 2008;7(5):391–9.
- Qureshi AI. Antihypertensive Treatment of Acute Cerebral Hemorrhage (ATACH): rationale and design. Neurocrit Care. 2007;6(1):56–66.
- 66. Rosenfeldt Z, Conklen K, Jones B, Ferrill D, Deshpande M, Siddiqui FM. Comparison of nicardipine with clevidipine in the management of hypertension in acute cerebrovascular diseases. J Stroke Cerebrovasc Dis. 2018;27(8):2067–73.
- Varelas PN, Abdelhak T, Corry JJ, James E, Rehman MF, Schultz L, et al. Clevidipine for acute hypertension in patients with subarachnoid hemorrhage: a pilot study. Int J Neurosci. 2014;124(3):192–8.
- 68. Graffagnino C, Bergese S, Love J, Schneider D, Lazaridis C, LaPointe M, et al. Clevidipine rapidly and safely reduces blood pressure in acute intracerebral hemorrhage: the ACCELERATE trial. Cerebrovasc Dis. 2013;36(3):173–80.
- 69. Ibanez B, James S, Agewall S, Antunes MJ, Bucciarelli-Ducci C, Bueno H, et al. 2017 ESC guidelines for the management of acute myocardial infarction in patients presenting with ST-segment elevation: the Task Force for the management of acute myocardial infarction in patients presenting with ST-segment elevation of the European Society of Cardiology (ESC). Eur Heart J. 2018;39(2):119–77.
- Sommariva S, Pongiglione B, Tarricone R. Impact of chemotherapyinduced nausea and vomiting on health-related quality of life and

resource utilization: a systematic review. Crit Rev Oncol Hematol. 2016;99:13–36.

- Aapro M. CINV: still troubling patients after all these years. Support Care Cancer. 2018;26(Suppl 1):5–9.
- Lohr L. Chemotherapy-induced nausea and vomiting. Cancer J. 2008;14(2):85–93.
- Hesketh PJ. Chemotherapy-induced nausea and vomiting. N Engl J Med. 2008;358(23):2482–94.
- Janelsins MC, Tejani MA, Kamen C, Peoples AR, Mustian KM, Morrow GR. Current pharmacotherapy for chemotherapy-induced nausea and vomiting in cancer patients. Expert Opin Pharmacother. 2013;14(6):757–66.
- Vig S, Seibert L, Green MR. Olanzapine is effective for refractory chemotherapy-induced nausea and vomiting irrespective of chemotherapy emetogenicity. J Cancer Res Clin Oncol. 2014;140(1):77–82.
- Ettinger DS, Berger MJ, Barbour S, Bergsbaken J, Brandt D, Brown GE, et al. NCCN Guidelines: antiemesis, version 2.2020 2020. Available from: https://www.nccn.org/professionals/physician_gls/pdf/antiemesis.pdf
- 77. Grunberg SM, Warr D, Gralla RJ, Rapoport BL, Hesketh PJ, Jordan K, et al. Evaluation of new antiemetic agents and definition of antineoplastic agent emetogenicity--state of the art. Support Care Cancer. 2011;19(Suppl 1):S43–7.
- Hesketh PJ, Kris MG, Basch E, Bohlke K, Barbour SY, Clark-Snow RA, et al. Antiemetics: American society of clinical oncology clinical practice guideline update. J Clin Oncol. 2017;35(28): 3240–61.
- 79. Roila F, Molassiotis A, Herrstedt J, Aapro M, Gralla RJ, Bruera E, et al. 2016 MASCC and ESMO guideline update for the prevention of chemotherapy- and radiotherapy-induced nausea and vomiting and of nausea and vomiting in advanced cancer patients. Ann Oncol. 2016;27(Suppl 5):v119–v33.
- Jordan K, Schmoll HJ, Aapro MS. Comparative activity of antiemetic drugs. Crit Rev Oncol Hematol. 2007;61(2):162–75.
- Sharma R, Tobin P, Clarke SJ. Management of chemotherapyinduced nausea, vomiting, oral mucositis, and diarrhoea. Lancet Oncol. 2005;6(2):93–102.
- Srivastava M, Brito-Dellan N, Davis MP, Leach M, Lagman R. Olanzapine as an antiemetic in refractory nausea and vomiting in advanced cancer. J Pain Symptom Manag. 2003;25(6): 578–82.
- Inoue T, Kimura M, Uchida J, Nishino K, Kumagai T, Taniguchi J, et al. Aprepitant for the treatment of breakthrough chemotherapyinduced nausea and vomiting in patients receiving moderately emetogenic chemotherapy. Int J Clin Oncol. 2017;22(3):600–4.
- 84. Jones JM, Qin R, Bardia A, Linquist B, Wolf S, Loprinzi CL. Antiemetics for chemotherapy-induced nausea and vomiting occurring despite prophylactic antiemetic therapy. J Palliat Med. 2011;14(7):810–4.
- 85. Navari RM, Nagy CK, Gray SE. The use of olanzapine versus metoclopramide for the treatment of breakthrough chemotherapyinduced nausea and vomiting in patients receiving highly emetogenic chemotherapy. Support Care Cancer. 2013;21(6):1655–63.
- 86. Chiu L, Chiu N, Chow R, Zhang L, Pasetka M, Stinson J, et al. Olanzapine for the prophylaxis and rescue of chemotherapyinduced nausea and vomiting (CINV): a retrospective study. Ann Palliat Med. 2016;5(3):172–8.
- 87. Nakagaki M, Barras M, Curley C, Butler JP, Kennedy GA. A randomized trial of olanzapine versus palonosetron versus infused ondansetron for the treatment of breakthrough chemotherapyinduced nausea and vomiting in patients undergoing hematopoietic stem cell transplantation. Support Care Cancer. 2017;25(2):607–13.
- Mehta RL, Awdishu L, Davenport A, Murray PT, Macedo E, Cerda J, et al. Phenotype standardization for drug-induced kidney disease. Kidney Int. 2015;88(2):226–34.

- Malyszko J, Kozlowska K, Kozlowski L, Malyszko J. Nephrotoxicity of anticancer treatment. Nephrol Dial Transplant. 2017;32(6):924–36.
- Kintzel PE. Anticancer drug-induced kidney disorders. Drug Saf. 2001;24(1):19–38.
- Schuchter LM, Hensley ML, Meropol NJ, Winer EP. 2002 update of recommendations for the use of chemotherapy and radiotherapy protectants: clinical practice guidelines of the American Society of Clinical Oncology. J Clin Oncol. 2002;20(12):2895–903.
- Santos MLC, de Brito BB, da Silva FAF, Botelho A, de Melo FF. Nephrotoxicity in cancer treatment: an overview. World J Clin Oncol. 2020;11(4):190–204.
- Izzedine H, Perazella MA. Anticancer drug-induced acute kidney injury. Kidney Int Rep. 2017;2(4):504–14.
- 94. Uzawa K, Kasamatsu A, Saito T, Kita A, Sawai Y, Toeda Y, et al. Growth suppression of human oral cancer cells by candidate agents for cetuximab-side effects. Exp Cell Res. 2019;376(2):210–20.
- Liu W, Qdaisat A, Soliman PT, Ramondetta L, Lopez G, Narayanan S, et al. Hypomagnesemia and survival in patients with ovarian cancer who received chemotherapy with carboplatin. Oncologist. 2019;24(6):e312–e7.
- Rosner MH, Capasso G, Perazella MA. Acute kidney injury and electrolyte disorders in the critically ill patient with cancer. Curr Opin Crit Care. 2017;23(6):475–83.
- Berghmans T. Hyponatremia related to medical anticancer treatment. Support Care Cancer. 1996;4(5):341–50.
- Liamis G, Filippatos TD, Elisaf MS. Electrolyte disorders associated with the use of anticancer drugs. Eur J Pharmacol. 2016;777:78–87.
- Kraft MD, Btaiche IF, Sacks GS, Kudsk KA. Treatment of electrolyte disorders in adult patients in the intensive care unit. Am J Health Syst Pharm. 2005;62(16):1663–82.
- Schrag D, Chung KY, Flombaum C, Saltz L. Cetuximab therapy and symptomatic hypomagnesemia. J Natl Cancer Inst. 2005;97(16):1221–4.
- Buckley JE, Clark VL, Meyer TJ, Pearlman NW. Hypomagnesemia after cisplatin combination chemotherapy. Arch Intern Med. 1984;144(12):2347–8.
- Lajer H, Daugaard G. Cisplatin and hypomagnesemia. Cancer Treat Rev. 1999;25(1):47–58.
- Schilsky RL, Anderson T. Hypomagnesemia and renal magnesium wasting in patients receiving cisplatin. Ann Intern Med. 1979;90(6):929–31.
- Blachley JD, Hill JB. Renal and electrolyte disturbances associated with cisplatin. Ann Intern Med. 1981;95(5):628–32.
- 105. Shahab I, Patterson W. Renal and electrolyte abnormalities due to chemotherapy. In: Perry M, editor. The chemotherapy source book. 4th ed. Philadelphia: Lippincott Williams & Wilkins; 2001. p. 223–44.
- Lyman NW, Hemalatha C, Viscuso RL, Jacobs MG. Cisplatininduced hypocalcemia and hypomagnesemia. Arch Intern Med. 1980;140(11):1513–4.
- 107. Ma Y, Hou L, Yu F, Lu G, Qin S, Xie R, et al. Incidence and physiological mechanism of carboplatin-induced electrolyte abnormality among patients with non-small cell lung cancer. Oncotarget. 2017;8(11):18417–23.
- 108. Hamdi T, Latta S, Jallad B, Kheir F, Alhosaini MN, Patel A. Cisplatin-induced renal salt wasting syndrome. South Med J. 2010;103(8):793–9.
- 109. Gilbar PJ, Richmond J, Wood J, Sullivan A. Syndrome of inappropriate antidiuretic hormone secretion induced by a single dose of oral cyclophosphamide. Ann Pharmacother. 2012;46(9):e23. https://doi.org/10.1345/aph.1R2961.
- Bruining DM, van Roon EN, de Graaf H, Hoogendoorn M. Cyclophosphamide-induced symptomatic hyponatraemia. Neth J Med. 2011;69(4):192–5.

- 111. Hammond IW, Ferguson JA, Kwong K, Muniz E, Delisle F. Hyponatremia and syndrome of inappropriate anti-diuretic hormone reported with the use of Vincristine: an over-representation of Asians? Pharmacoepidemiol Drug Saf. 2002;11(3):229–34.
- 112. Stohr W, Paulides M, Bielack S, Jurgens H, Treuner J, Rossi R, et al. Ifosfamide-induced nephrotoxicity in 593 sarcoma patients: a report from the Late Effects Surveillance System. Pediatr Blood Cancer. 2007;48(4):447–52.
- 113. Stintzing S, Fischhaber D, Mook C, Modest DP, Giessen C, Schulz C, et al. Clinical relevance and utility of cetuximab-related changes in magnesium and calcium serum levels. Anti-Cancer Drugs. 2013;24(9):969–74.
- 114. Vincenzi B, Santini D, Galluzzo S, Russo A, Fulfaro F, Silletta M, et al. Early magnesium reduction in advanced colorectal cancer patients treated with cetuximab plus irinotecan as predictive factor of efficacy and outcome. Clin Cancer Res. 2008;14(13):4219–24.
- 115. Chen P, Wang L, Li H, Liu B, Zou Z. Incidence and risk of hypomagnesemia in advanced cancer patients treated with cetuximab: a meta-analysis. Oncol Lett. 2013;5(6):1915–20.
- 116. Milionis HJ, Liamis GL, Elisaf MS. The hyponatremic patient: a systematic approach to laboratory diagnosis. CMAJ. 2002;166(8):1056–62.
- 117. Spasovski G, Vanholder R, Allolio B, Annane D, Ball S, Bichet D, et al. Clinical practice guideline on diagnosis and treatment of hyponatraemia. Intens Care Med. 2014;40(3):320–31.
- 118. Lenz HJ. Management and preparedness for infusion and hypersensitivity reactions. Oncologist. 2007;12(5):601–9.
- 119. Kanji S, Chant C. Allergic and hypersensitivity reactions in the intensive care unit. Crit Care Med. 2010;38(6 Suppl):S162–8.
- Syrigou E, Makrilia N, Koti I, Saif MW, Syrigos KN. Hypersensitivity reactions to antineoplastic agents: an overview. Anti-Cancer Drugs. 2009;20(1):1–6.
- 121. Lieberman P. Biphasic anaphylactic reactions. Ann Allergy Asthma Immunol. 2005;95(3):217–26; quiz 26, 58.
- 122. Shaker MS, Wallace DV, Golden DBK, Oppenheimer J, Bernstein JA, Campbell RL, et al. Anaphylaxis-a 2020 practice parameter update, systematic review, and Grading of Recommendations, Assessment, Development and Evaluation (GRADE) analysis. J Allergy Clin Immunol. 2020;145(4):1082–123.
- 123. Castells Guitart MC. Rapid drug desensitization for hypersensitivity reactions to chemotherapy and monoclonal antibodies in the 21st century. J Investig Allergol Clin Immunol. 2014;24(2):72–9.
- Worth A, Soar J, Sheikh A. Management of anaphylaxis in the emergency setting. Expert Rev Clin Immunol. 2010;6(1):89–100.
- 125. Simons FE. Pharmacologic treatment of anaphylaxis: can the evidence base be strengthened? Curr Opin Allergy Clin Immunol. 2010;10(4):384–93.
- Simons FE, Gu X, Simons KJ. Epinephrine absorption in adults: intramuscular versus subcutaneous injection. J Allergy Clin Immunol. 2001;108(5):871–3.
- Simons KJ, Simons FE. Epinephrine and its use in anaphylaxis: current issues. Curr Opin Allergy Clin Immunol. 2010;10(4):354–61.
- 128. Campbell RL, Bellolio MF, Knutson BD, Bellamkonda VR, Fedko MG, Nestler DM, et al. Epinephrine in anaphylaxis: higher risk of cardiovascular complications and overdose after administration of intravenous bolus epinephrine compared with intramuscular epinephrine. J Allergy Clin Immunol Pract. 2015;3(1):76–80.
- 129. Lieberman P, Nicklas RA, Oppenheimer J, Kemp SF, Lang DM, Bernstein DI, et al. The diagnosis and management of anaphylaxis practice parameter: 2010 update. J Allergy Clin Immunol. 2010;126(3):477–80, e1–42.
- Goolsby TV, Lombardo FA. Extravasation of chemotherapeutic agents: prevention and treatment. Semin Oncol. 2006;33(1):139–43.
- Ener RA, Meglathery SB, Styler M. Extravasation of systemic hemato-oncological therapies. Ann Oncol. 2004;15(6):858–62.

- Kreidieh FY, Moukadem HA, El Saghir NS. Overview, prevention and management of chemotherapy extravasation. World J Clin Oncol. 2016;7(1):87–97.
- 133. Jackson-Rose J, Del Monte J, Groman A, Dial LS, Atwell L, Graham J, et al. Chemotherapy extravasation: establishing a national benchmark for incidence among cancer centers. Clin J Oncol Nurs. 2017;21(4):438–45.
- 134. Pérez Fidalgo JA, García Fabregat L, Cervantes A, Margulies A, Vidall C, Roila F. Management of chemotherapy extravasation: ESMO-EONS clinical practice guidelines. Ann Oncol. 2012;23(Suppl 7):vii167–73.
- Wickham R, Engelking C, Sauerland C, Corbi D. Vesicant extravasation part II: evidence-based management and continuing controversies. Oncol Nurs Forum. 2006;33(6):1143–50.
- Vidall C, Roe H, Dougherty L, Harrold K. Dexrazoxane: a management option for anthracycline extravasations. Br J Nurs. 2013;22(17):S6–12.
- Schulmeister L. Extravasation management. Semin Oncol Nurs. 2007;23(3):184–90.
- Patel JS, Krusa M. Distant and delayed mitomycin C extravasation. Pharmacotherapy. 1999;19(8):1002–5.
- Al-Benna S, O'Boyle C, Holley J. Extravasation injuries in adults. ISRN Dermatol. 2013;2013:856541.
- 140. Mouridsen HT, Langer SW, Buter J, Eidtmann H, Rosti G, de Wit M, et al. Treatment of anthracycline extravasation with Savene (dexrazoxane): results from two prospective clinical multicentre studies. Ann Oncol. 2007;18(3):546–50.
- 141. Fontaine C, Noens L, Pierre P, De Grève J. Savene® (dexrazoxane) use in clinical practice. Support Care Cancer. 2012;20(5):1109–12.
- 142. Olver IN, Aisner J, Hament A, Buchanan L, Bishop JF, Kaplan RS. A prospective study of topical dimethyl sulfoxide for treating anthracycline extravasation. J Clin Oncol. 1988;6(11):1732–5.
- Schrijvers DL. Extravasation: a dreaded complication of chemotherapy. Ann Oncol. 2003;14(Suppl 3):iii26–30.

- 144. Bertelli G, Dini D, Forno GB, Gozza A, Silvestro S, Venturini M, et al. Hyaluronidase as an antidote to extravasation of Vinca alkaloids: clinical results. J Cancer Res Clin Oncol. 1994;120(8):505–6.
- 145. Sauerland C, Engelking C, Wickham R, Corbi D. Vesicant extravasation part I: mechanisms, pathogenesis, and nursing care to reduce risk. Oncol Nurs Forum. 2006;33(6):1134–41.
- Stanford BL, Hardwicke F. A review of clinical experience with paclitaxel extravasations. Support Care Cancer. 2003;11(5):270–7.
- 147. Sibaud V, Lebœuf NR, Roche H, Belum VR, Gladieff L, Deslandres M, et al. Dermatological adverse events with taxane chemotherapy. Eur J Dermatol. 2016;26(5):427–43.
- 148. Jordan K, Behlendorf T, Mueller F, Schmoll HJ. Anthracycline extravasation injuries: management with dexrazoxane. Ther Clin Risk Manage. 2009;5(2):361–6.
- 149. Maly C, Fan KL, Rogers GF, Mitchell B, Amling J, Johnson K, et al. A primer on the acute management of intravenous extravasation injuries for the plastic surgeon. Plast Reconstr Surg Glob Open. 2018;6(4):e1743.
- Dorr RT. Antidotes to vesicant chemotherapy extravasations. Blood Rev. 1990;4(1):41–60.
- 151. Boschi R, Rostagno E. Extravasation of antineoplastic agents: prevention and treatments. Pediatr Rep. 2012;4(3):e28.
- 152. Laurie SW, Wilson KL, Kernahan DA, Bauer BS, Vistnes LM. Intravenous extravasation injuries: the effectiveness of hyaluronidase in their treatment. Ann Plast Surg. 1984;13(3):191–4.
- 153. Bozkurt AK, Uzel B, Akman C, Ozgüroğlu M, Molinas MN. Intrathoracic extravasation of antineoplastic agents: case report and systematic review. Am J Clin Oncol. 2003;26(2):121–3.
- 154. Mas V, Simon AL, Presedo A, Mallet C, Ilharreborde B, Jehanno P. Upper limb extravasation of cytotoxic drugs: results of the saline washout technique in children. J Child Orthop. 2020;14(3):230–5.
- Goutos I, Cogswell LK, Giele H. Extravasation injuries: a review. J Hand Surg Eur Vol. 2014;39(8):808–18.

Treatment Toxicity: Radiation

Thomas J. FitzGerald, Maryann Bishop-Jodoin, Fran Laurie, Eric C. Ko, Allison Sacher, Gregory A. Volturo, Daniel C. Han, Paul S. Rava, and Eric W. Dickson

Introduction

Exposure to radiation can result from diagnosis and primary management of a malignancy, or it can be an unintended consequence of nuclear accidents or an intentional act of terrorism. More than 60% of cancer patients will have radiation therapy as part of their primary treatment, and the accelerated risk of nuclear events increases our collective need for education of health-care providers in the knowledge and evaluation of radiation exposure and injury [1-8]. Although effects on tissue may not be visible or clinically apparent during an emergency evaluation, exposure is an important aspect to the medical history of the patient as it leaves an invisible clinical footprint which may be relevant to medical situations at a later time point, even decades after exposure. Intentional exposure is usually well documented with dosevolume precision in the radiation oncologic treatment record. However, at the time of an unanticipated emergency department visit, hospital record documentation is often limited to a few words describing treatment total dose and volume in a brief qualitative manner as the shadow radiation therapy record or radiation digital health record is often not directly linked to the hospital informatics systems. Therefore, information valuable to the emergency health-care team may be cursory, incomplete, or even inaccurate if obtained from a service unfamiliar with radiation treatment and exposure to normal tissue.

M. Bishop-Jodoin · F. Laurie

G. A. Volturo · E. W. Dickson

© Springer Nature Switzerland AG 2021

Unintentional exposure is more challenging to document and often limited to mathematical models of duration and distance from the primary incident, as victims are often unmonitored. While the models may be helpful, they often can be less accurate, especially in computation of the integral or total body dose [1, 6, 8]. Fluoroscopy during interventional radiology/cardiology procedures can lead to a surprisingly high-radiation dose to underlying structures that is often poorly documented, again comprising a relatively hidden risk in patient care [2–4].

With an increasing number of cancer survivors, including transition of survivors from pediatric to adult physicians, there is a developing knowledge gap at the primary care and emergency medicine levels concerning both acute and late effects of radiation exposure and how these interrelate with patient health care in the acute care setting. It is incumbent on the radiation oncologic community and radiation exposure experts to improve documentation and communication to health-care staff in order to better prepare patients and physicians for identifying contributing factors and strategies of short- and long-term normal tissue-driven processes that affect patients after radiation treatment and exposure.

Acute Radiation Toxicity: Unintended Exposure

Normal Tissue Toxicity

Acute toxicity from radiation exposure can be divided into acute (up to 90 days from exposure), subacute (from 90 days to 2 years from exposure), and chronic or late (>2 years from exposure) phases of injury. Although acute intentional injuries are traditionally managed by the responsible treating physicians during primary management of a malignancy, accelerated normal tissue damage that affects hydration balance and nutrition can often require evaluation by emergency services for triage and disposition. Unintentional exposure uniformly requires evaluation by emergency services with



T. J. FitzGerald (⊠) · E. C. Ko · A. Sacher · D. C. Han · P. S. Rava Department of Radiation Oncology, UMass Memorial Health Care, University of Massachusetts Medical School, Worcester, MA, USA e-mail: TJ.FitzGerald@umassmemorial.org

Department of Radiation Oncology, University of Massachusetts Medical School, Worcester, MA, USA

Department of Emergency Medicine, UMass Memorial Health Care, University of Massachusetts Medical School, Worcester, MA, USA

T. J. FitzGerald et al.

appropriate support from experts trained in managing acute radiation effects and radiation safety officers trained in assessing the nature of the exposure and possible radiation dose received by the victims. This would also include the evaluation of risk to personnel caring for the victim. The acute phase of injury can affect many cell systems, including toxicity to tissues of both limited and rapid self-renewal potential. These include, but are not limited to, the central nervous system, bone marrow, skin, and mucosal surfaces lining the head/neck and gastrointestinal system.

Knowledge of the effects of radiation damage to the body has been acquired through animal models as well as events in history documenting human exposure. These events include atomic bomb survivors and people affected through unintended nuclear fallout and nuclear accidents. Symptoms associated with unintended radiation exposure vary with the severity of the exposure. At very high single-fraction total body doses (>10 Gy), death will occur through cerebrovascular syndrome in spite of support within 24-48 h. The syndrome is due to uncontrollable swelling within the central nervous system associated with compromise of all neuromuscular processes. At total body doses of 5-12 Gy, death without support will occur in 1-2 weeks due to denudation and destruction of the gastrointestinal system associated with profound fluid loss and diarrhea. These cells have a self-renewal capacity measured within a few days; thus, a single total body dose of 10 Gy will eliminate a large portion of the stem cells within the gastrointestinal crypts. Although this dose does not affect differentiated adult cells, the exposure eliminates the self-renewal potential of the stem cell; therefore, the mucosal surface of the gastrointestinal tract becomes denuded with no barrier for fluid and blood loss within a short period of time, measured in days. At total body exposure doses of 2-5 Gy, death will occur from damage to the hematopoietic system with primary damage to progenitor cells inhibiting self-renewal. Lymphocytes may die an intermitotic death, and this finding may be a surrogate biomarker for acute exposure within the first few hours to days of an incident [1, 5–8]. However, by day 30, most circulating blood elements are depleted with death often attributed to infection. The term, LD (lethal dose) 50/30, is borrowed from our pharmacology colleagues and reflects the LD of an agent that will cause 50% mortality in 30 days. Although radiation is not a drug, the LD 50/30 is now generally thought to be 5 Gy with modern hospital support [1].

After exposure, victims will develop symptoms consistent with a radiation syndrome that can be visible as early as 15 minutes from the time of exposure [1, 6, 8]. The reaction may last several days until symptoms merge with other events associated with the exposure. Symptoms are generally gastrointestinal and neuromuscular. At lower doses, victims experience anorexia, nausea, and vomiting associated with lassitude. The degree of symptoms is commensurate with dose. At higher doses, patients can experience severe diarrhea, fever, and hypotension suggesting more immediate toxicity including more pronounced neural damage. Usually at low doses, the prodromal phase is followed by a latent period where the victim may appear and feel well for a period of days to weeks. At that point gastrointestinal and hematopoietic damage becomes more evident and requires intervention [1, 5, 6, 8].

If the total body exposure is less than 4–5 Gy, the majority of experts recommend no immediate intervention other than symptomatic treatment as needed. This would include hydration and antiemetic therapy for nausea/vomiting. Antibiotics can be given for infection as needed. If the exposure is greater than 5 Gy, then death associated with the hematopoietic syndrome becomes a real concern. Intervention with isolation and barrier nursing with appropriate blood product support may improve survival. Experience from Chernobyl suggests that efforts to limit infection, bleeding, and physical trauma during the time of blood count nadir may improve the LD 50/30 to and possibly beyond 7 Gy. To the best of current knowledge, no human being has survived a single total body exposure beyond 10 Gy. The use of bone marrow transplant in this setting remains controversial with strong advocates on both sides of the question [7].

Injuries to the gastrointestinal system and hematopoietic systems may be accompanied by dermal injury. Often in particle exposure including low-energy photons, there is significant asymmetry in dose as often the event is triggered by an accident involving the hands. Dermal injuries can be primitive biomarkers for dose with epilation/erythema at doses of 3–6 Gy and wet desquamation, bullae, ulceration, and necrosis seen at higher doses [1, 9]. These injuries can be life-threatening due in part to concomitant infection and should be managed with the same support offered to burn victims.

In the triage of victims with unintended exposure, it is most important to develop as accurate an assessment of dose as possible. Health-care workers will likely be monitored; however, the general public will not, and therefore experts trained in radiation exposure and dose assessment are crucial in the early phase of the evaluation. There are several basic tools to use as part of the initial evaluation. The time to emesis decreases with increasing radiation dose. The rapid onset of nausea and vomiting suggests a higher exposure. A decline in lymphocyte count can be associated with an exposure dose of best relative estimate occurring within 48 h of exposure. This is often hampered because the preexposure lymphocyte count is unknown. If a laboratory for cytogenetic evaluation is available, another surrogate evaluation is to assess the number of chromosomal aberrations in peripheral lymphocytes when they are stimulated to divide. This technique has more value at lower doses as lymphocytes die quickly at high doses [1, 7, 8].

There have been numerous nuclear accidents over the past 50 years with people exposed to total body or partial body radiation. These include nuclear events as well as unintentional exposure to victims unaware of the immediate risk [6, 8]. The Medical Sciences Division of the Oak Ridge Institute for Science and Education operates a Radiation Emergency Assistance Center for the US Department of Energy. The center is a 24-h consultation service with medical and health physics support for issues associated with radiation events and exposure. Resources include expertise for radiation dose assessment, computation of dose from radionucleotides, and laboratory facilities for dose assessment. The 24-h emergency telephone number is 865.576.3131, and the website is http://www.orau.gov/reacts.

Since the development of nuclear weapons, there has been a keen scientific interest in identifying chemical compounds that can protect normal tissues from the effects of radiation exposure. Radiation protectors are elements that are given prior to exposure or in some cases, shortly thereafter, to limit the effect of exposure on normal tissue. Radiation mitigators are compounds that have the potential of influencing the effect and impact of the exposure. Therapeutic compounds are applied once the injury has occurred. There have been more than 4000 compounds synthesized to address this point. Sulfhydryl compounds (SHs) have been shown to be effective radioprotectors with the simplest compound being cysteine, which contains a natural amino acid. It was shown in 1949 that this compound could protect animals from lethal doses of total body radiation if injected or injected in large amounts (equivalent of 150 mg/ kg) [10, 11]. The toxicity of sulfhydryl compounds can be limited by the addition of a phosphate group [10]. Once the compound becomes intracellular, it loses the phosphate group, and the compound is thought to serve as a free radical scavenger limiting intracellular damage. The only compound approved by the US Food and Drug Administration (FDA) is amifostine (WR-2721). It is sold as Ethyol and has been used to prevent xerostomia in patients undergoing radiation therapy for head/neck carcinoma [12]. Amifostine has been used in several clinical trials evaluating effectiveness in protecting multiple mucosal surfaces as well as a protectant for pulmonary injury in patients undergoing total body radiation therapy as part of bone marrow transplant [12]. In a Radiation Therapy Oncologic Group (RTOG) clinical trial, amifostine was associated with an improvement in patient assessment of mouth dryness and swallowing [12, 13]. In this trial, there was no difference in tumor control between patients receiving amifostine or placebo. Citrin and colleagues [10] have identified nitroxides as agents for radioprotection in clinical development. Stable nitroxide free radicals and their specific electron reduction products, hydroxylamines, protect cells when exposed to oxidative stress; therefore, compounds such as these are under evaluation. Other antioxidants, including alpha-tocopherol and beta-carotene, have not yet been shown to be of clinical benefit [14, 15]. Simultaneous tumor protection has been a concern in the clinical use of radioprotectors and is the reason these compounds have not been easily integrated into clinical management. Investigators have explored the use of intracellular superoxide dismutase (SOD) using gene therapy vectors to enhance the intracellular component of SOD to limit damage caused by superoxide radicals [16, 17].

Mitigators are compounds that can limit damage associated with radiation exposure prior to the clinical manifestation of both acute and late toxicities of radiation exposure and treatment. These compounds are generally thought to influence the metabolic cascade of events that occur after exposure and in turn limit radiation-associated damage. To date, most of the compounds are cytokines and growth factors directed to stimulate stem cell proliferation and balance the inhibition of stem cell growth induced by radiation to the hematopoietic and GI systems. These include granulocyte colony-stimulating factor (G-CSF) and keratinocyte growth factor (KGF) [18]. These factors contribute to many aspects of cell recovery. KGF has a positive influence in the recovery of mucosal surfaces during the acute phase of toxicity as well as limits the late effects of radiotherapy, including xerostomia [10]. Mitigators of late toxicity are largely directed to limit fibrosis, which is thought to be a primary factor in late pulmonary injury and other tissues of more limited selfrenewal potential [10, 14–17, 19–23]. The primary target for this strategy is thought to be transforming growth factor beta (TGF B), which appears to play an important role in the development of fibrosis associated with radiation [24-26]. Accordingly, many compounds in development to prevent late effects either directly or indirectly target the TGF B signaling pathway including receptor inhibition [24–26]. Tumor protection is also a concern in the evaluation of treatments associated with this parallel pathway.

Investigators at the University of Massachusetts Medical School have evaluated the use of interleukin-1 alpha (IL-1 α) as a mitigator to dermal damage associated with radiation exposure. Interleukin-1 (IL-1) inhibits neutrophil infiltration into the initial inflammatory response to radiation damage. Dermal injury was induced with electron particle therapy. Knockout mice deficient in IL-1a or the IL-1 receptor demonstrated both decreased dermal injury and more rapid healing suggesting the importance of this cytokine in the generation of radiation-associated skin damage. Neutrophil inhibition generated subsequent to radiation-induced tissue injury influences the pathogenesis of radiodermatitis. In a separate group of experiments, investigators from the same institution demonstrated that hyperspectral optical imaging (HSI) can demonstrate both acute and late oxygenation and perfusion changes in dermal tissue with changes occurring as early as 12 h after radiation exposure using a strontium-90

applicator [27, 28]. Imaging changes in oxygenation and perfusion predated clinical visible skin change by 14 days [28]. Unpublished data sets from this group as part of a human IRB clinical trial in breast cancer patients undergoing radiation therapy have shown that changes in imaging correlate very well with radiation dose and dose asymmetry in the treated volume.

In summary, with the increased risk of nuclear radioterrorism and increased radiation exposure identified during air and space travel, there is a renewed sense of urgency to better define and refine our response to a nuclear event. It is important to acknowledge that dose and particle contamination become essential points during the initial triage of the patient. Potential exposure to first responders must be assessed. The best supportive care has the potential of improving patient survival, which may include hydration and blood product support. There is renewed interest in developing a targeted pharmacologic response to both protect and mitigate issues surrounding radiation exposure [29].

Normal Tissue Effects of Radiation Therapy

Often oncologic patients under active treatment are seen in evaluation by emergency physicians, particularly during non-primary clinic times including evenings and weekends. In large academic centers, this practice is under change as outpatient weekend service is becoming more important to clinical oncologic as more primary patient care management, including bone marrow transplantation, moves to the outpatient setting. Nevertheless, often emergency personnel are involved with patient care matters that are directly or indirectly associated with the management of the malignancy or the sequelae of management including triage of acute care problems that may be related in part to previous therapy delivered years and decades in the past. In the following sections, we will address sequelae of management and how this influences patient care for modern emergency medicine [30].

Management of Acute Effects of Radiation Therapy

Sequelae to normal tissue are mostly attributed to cell death from radiation therapy. The balance between stem cell development and cell death is driven in a large part by tissue organization, stem cell proliferation, stem cell number, and cytokine response for growth stimulus. Acute effects from therapeutic radiation therapy are associated with cell systems that have rapid self-renewal potential including bone marrow progenitors and mucosal surfaces. These sequelae are driven by several factors including the concomitant use of cytotoxic chemotherapy and the volume of mucosal tissue in the radiation therapy treatment field. Acute effects generally occur during the course of radiation therapy and can affect multiple organ systems largely associated with mucosal surfaces including the skin, head/neck, gastrointestinal tract, and bone marrow. Acute effects are exacerbated by the use of chemotherapeutic and/or molecularly targeted small molecule agents delivered both before and during radiation therapy. Although not well validated through mechanism, patients can experience dramatic acute effects from low-dose radiation therapy to the skin and mucosa if they have received prior sensitizing medications including low-dose chemotherapy for autoimmune disease (methotrexate for rheumatoid arthritis) and selected antibiotics (tetracycline) [9]. The phenomenon, referred to as radiation recall, can even be seen in patients who received medications years in the past [9]. Although acute effects mostly impact tissues of rapid selfrenewal potential, there can be selected circumstances where near-immediate changes occur that are often not anticipated by the primary providers of care. Patients treated in the head and neck region can experience swelling of the parotid glands with 24 h of exposure of 200 cGy. The adventitia of the parotid gland is tight with limited capacity for expansion from swelling; therefore, rare patients can experience severe pain and discomfort from low-dose therapy. There is often a need to urgently treat patients with significant tumor burdens, particularly in the mediastinum. Radiation to tumors of the mediastinum particularly sensitive to treatment (lymphoma, small cell lung cancer, or germ cell neoplasm) can trigger both nausea and metabolic crisis (hyperkalemia, hypercalcemia, etc.) from rapid tumor lysis. Symptomatic treatment including fluids and medication to counter metabolic by-products is essential for a good outcome.

Acute effects to tissues of rapid self-renewal potential are influenced by total radiation dose, daily treatment dose (fractionation), and volume of tissue treated. This information is often not immediately available to emergency departments when patients present for evaluation as specific treatment documentation is often in the department shadow record and not directly integrated into electronic health-care records. The volume of the treatment target influences the number of stem cells directly affected from daily treatment. Daily treatment dose also influences injury to stem cells. Hence, total and daily doses as well as target volume are all directly related to the development of sequelae from the treatment. Specific acute injuries to organ systems are discussed in the following section.

Skin

The epidermis is the site of many acute reactions to radiation exposure. The dermal stem cells abut the basement membrane and are the active proliferating cell component covered by layers of keratinized cells, which are desquamated. The stem cells are the target for injury. The time for dermal cell division and migration is between 14 and 21 days depending on the area of the body under evaluation. Single doses of 5 Gy will generate early erythema followed by vasodilation, fluid exudation, cellular migration, and loss of proteins and other constituents of plasma products [9, 31, 32]. Investigators have shown that this process can be identified on hyperspectral imaging within 12 h of exposure with evidence that the evaluation on imaging can be dose specific in spite of the fact that clinical expression of change may not become apparent for 2-3 weeks [27, 28]. Fluoroscopy procedures use orthovoltage (low-energy) X-rays that deliver higher percentages of radiation dose to the skin surface [2, 3]. Complicated procedures in interventional radiology requiring significant fluoroscopy time can create acute dermal injury even in the modern era as acute injuries are influenced by fractionation (daily dose) and total dose. With hypofractionation protocols using high daily dose with compressed treatment schedules including stereotactic therapy for the lung and liver, we are again witnessing injuries to the skin and soft tissue that were traditionally seen in a historical context [33]. The treatment for acute injury is driven in a large part by radiation dose and treatment volume. Modern accelerators deliver the majority of the radiation dose below the skin surface; therefore, with traditional fractionation, it is unusual to have patients demonstrate significant dermal sequela with conservative measures, including various skin creams and ointments. Radiation beams resonate on skin surfaces within dermal folds; therefore, these intertriginous areas are more vulnerable to injury during treatment. Hypofractionation protocols may deliver a higher-dose fractionation to dermal surfaces if treatment planning is not optimal. There are reported soft tissue injuries to the skin during stereotactic body radiosurgery when immobilization devices unintentionally functioned as bolus devices augmenting radiation dose to skin surfaces [33]. As information matures on molecularly targeted therapies, there is increasing evidence of skin toxicity to multiple new agents including EGFR inhibitors (rash 2-4 weeks into therapy), BRAF inhibitors (rash/photosensitivity), BCR-ABL inhibitors (keratosis pilaris/maculopapular rash), and m-TOR inhibitors (rash/pruritus) [10].

Hematopoietic System

The effects on the hematopoietic system are driven by the volume of the bone marrow and lymphoid system treated, previous chemotherapy and radiation treatment, as well as radiation dose. Age of the patient also influences bone marrow injury as there is asymmetry of location in marrow locations between children and adults. Total body exposure will result in a near-immediate decrease in circulating B and T lymphocytes, and a total body dose of 3.0–4.0 Gy likely

inhibits the ability to respond to new antigen stimuli. Most patients receive radiation treatment to a partial body, and usually partial organ volume that would have a limited effect on the immune response unless the patient is neutropenic from concurrent chemotherapy [34]. This permits migration of stem cells from outside of the therapy field to stabilize bone marrow function.

Gastrointestinal Tract

The mucosa of the tract has similar organization to skin tissues as stem cells reside at the basal layer and migrate to the surface at varied time points during their life cycle. In general, the cells that line segments of the gastrointestinal (GI) tract possess a shorter life span than their counterparts in the skin. The mucosa of the head and neck and large bowel self-renew every 2 weeks, the mucosa of the gastric region renews nearly every day, and the small intestine renews every 3 days. This explains, in part, why nausea from therapy directed to the gastric region and small bowel can be apparent very early postexposure.

Because the mucosal systems have rapid self-renewal potential, acute sequela from management can be substantial and is driven by radiation dose, daily treatment fraction, and volume of mucosa in the treatment field. By the second week of treatment, the mucosa of the head and neck becomes denuded with increasing pain. Secondary tissues including salivary glands and taste buds also display limited function driven in a large part by the volume of mucosa in the treatment field. By week 4, the mucosa will slough and be replaced by confluence of white cells and fibrin exudate. The impact on secondary tissues becomes more pronounced with severe xerostomia and loss of taste. This creates challenges with maintaining adequate dental hygiene and nutrition. Often patients treated to substantial mucosal volumes require supplemental nutrition for extended periods of time during and beyond treatment completion. These patients can be cured of their primary malignancy; therefore, adopting an aggressive approach to the management of the acute effects from the treatment is reasonable and medically appropriate. Symptomatic pain management with topical and enteral medications is an important management vehicle for care during this period of time [34].

The epidemiology of carcinoma of the esophagus has changed during the past 30 years in North America. Squamous cell carcinomas associated with alcohol and tobacco use have been replaced by primary adenocarcinomas, largely of the distal esophagus associated with gastric surface gland migration (Barrett's esophagus). This has been seen in multiple countries and is now the great majority of esophageal cancers. The mucosa of the esophagus will self-renew in a time frame similar to head and neck mucosa; therefore, 2 weeks into a treatment course, the patient will begin to develop swallowing discomfort related to treatment. Often these patients initially feel improved due to tumor response; however, mucosal denudation from treatment also becomes apparent during this time, and symptoms associated with dehydration and nutritional imbalance become more visible. Symptomatic management with pain medication and fluid support is an important adjunct during this period.

Treatment of the gastric mucosa can cause near-immediate nausea/vomiting due in a large part to the rapid self-renewal capacity of gastric stem cells. During a treatment course of radiation therapy, delayed gastric emptying can be observed due to edema in the bowel wall as well as the development of ulceration due to limited stem cell renewal capacity.

Early complications of the small and large bowel are similarly driven by radiation dose, treatment fractionation, volumes of bowel in the radiation therapy treatment field, and previous abdominal surgery. Previous abdominal surgery can result in adhesions which can fix segments of the bowel into a specific location, potentially exacerbating acute injury due to limitation of blood supply and repeated high-dose treatment [15, 35]. The small bowel absorbs protein, carbohydrate, fat, and water. If the mucosal surface is denuded, foodstuffs cannot be absorbed. Carbohydrate and fat function as a micelle and draw more water into the gastrointestinal tract resulting in increased bowel movement frequency and symptoms consistent with malabsorption. The large bowel mainly absorbs water; therefore, if generous segments of the large bowel are included in the therapy field, increased bowel frequency may occur with risk of dehydration and electrolyte loss often exacerbated with the concurrent chemotherapy. Although some of these issues can be anticipated and addressed through daily clinical care with fluid and electrolytes, often, these patients present for emergency department evaluation during evenings and weekends for symptom management. Therefore, up-to-date clinical information can often facilitate and support emergency services when needed including the risk of secondary infections associated with therapy. From a clinical perspective, patients being treated for recurrent disease often have more acute and potentially more serious sequela than patients being treated on an adjuvant basis. The root cause for this phenomenon is multifactorial, likely driven in part by tumor compromise of normal tissue function and vasculature prior to the initiation of therapy [34].

Subacute and Late (Delayed) Effects of Treatment

For both primary and emergency health-care providers, the late effects of cancer management can be less visible to casual observation but become a highly visible component to patient care years and decades after primary management. As patients are cured of their primary malignancy, secondary effects of therapy on normal tissue structure and function are now important for follow-up and preventative care as needed. As children treated for malignancy grow and become adults, adult physicians will need to embrace previous therapy as a significant component to their past medical history and treat accordingly. Acute effects of treatment are not always a predictive indicator of late effects, specifically patients without acute effects during the primary management phase of treatment remain at identifiable risk for late effects.

Usually acute effects of radiation management affect cells that have a rapid self-renewal capacity. Nearly every cell system in both adults and children is at risk for late effects driven in a large part by the total radiation dose and volume of normal tissue treated. Late effects are also influenced by daily treatment dose. Children remain at significant lifetime risk as growth and development of all cell systems is directly affected by therapy [15].

The relationship of chemotherapy to delayed effects from radiation therapy to normal tissue is less well described [15]. Although it is recognized in a qualitative manner that chemotherapy exacerbates acute effects of radiation therapy to tissues of rapid self-renewal potential, the impact of chemotherapy on late effects of treatment is not understood. To date, most radiation oncologists have not made adjustments in dose to normal tissue targets; however, our volumetric dose volume tools are now providing metrics through histogram analysis. The quantitative analysis of normal tissue effects in the clinic (OUANTEC) is a significant effort by radiation oncologists to define dose volumetrics for thresholds for normal tissue injury [36]. This effort reviewed most of the available and relevant published data to date and provides guidelines for physicians to follow for prevention of injury [36]. Because of modern computer metrics, the guidelines are driven by radiation dose and volume exposed to treatment. When treatment is developed through three-dimensional planning and executed through conformal delivery systems, including intensity modulation, normal tissue volumetrics are available for analysis through dose-volume histogram tools.

In this section, the late effects of radiotherapy will be described. This is important for emergency physicians as often this information is not available as part of the past medical history of the patient during an acute care evaluation and triage for advanced medical care. Improved knowledge of late effects will influence management and evaluation including the evaluation strategy in the acute care setting. Understanding these effects will improve patient care and evaluation moving forward.

Skin

Acute effects (described previously) of radiation injury to the skin generally resolve within 1 month of completion of therapy. With traditional fractionation to radiation doses of 7000 cGy, there can be thinning of the epidermis with deceptive prominence of the vascular pattern (telangiectasia) in the dermis. The degree of vascularity is decreased; however, thinning of the epidermis can make the vessels in the dermis appear more prominent [9]. With hypofractionation protocols now in clinical use, there can be more visible injury associated with varying degrees of fibrosis [33]. Hyperspectral imaging demonstrates that oxygenation is decreased, providing an explanation to limitations in wound healing when there are secondary injury and infection [10, 27]. Local immunity and moisture glands significantly diminish; therefore, injury to irradiated tissue can result in significant delay in healing with persistent ulceration and damage possibly requiring aggressive surgical care including hyperbaric oxygen therapy in extreme situations [9]. To date there has been little progress on specific therapies that can be applied to radiation injury once it is clinically apparent. The current standard of care is to follow optimal wound care strategies.

Skin tissue can also demonstrate a phenomenon termed *recall*. Chemotherapy agents and antibiotic agents can cause acute erythema and skin breakdown in areas that were previously irradiated. This can occur years after primary radiation therapy management.

Patients with autoimmune disorders including lupus and scleroderma may be vulnerable to accelerated fibrosis from traditional radiation therapy. This creates wound healing issues for secondary injury as well as limitations in function of organs requiring coordinated muscle function including the esophagus [9].

The mechanism of visible injury is due to damage to stem cells and the reticulum support elements of the skin. The repair becomes disorganized, and the injury to the dermis does not support the development of the epidermis as in an untreated skin. Secondary infections are more difficult to heal when identified in treated areas.

Bone Marrow

Although the primary focus of attention is with acute effects, the bone marrow can remain fragile years after therapy and vulnerable to various medications and external agents after radiation therapy. This is especially true for white cell elements and platelets. If there is a diminution in red cell count, this is usually not associated directly with radiation therapy and will require more extended evaluation for blood loss and anemia. Pancytopenia and bone marrow aplasia are becoming more common consequences of cancer therapy and often first identified in an acute care environment. This often influences choice and duration of treatment of infections and other late sequelae of management. Secondary blood dyscrasias including secondary blood malignancies are associated with primary management and are often first identified in the acute care setting [15, 34]. The role of transplant of both primary and support cell marrow elements to repopulate bone marrow remains under study.

Gastrointestinal Tract

There are substantial late effects of management to be considered in the acute care environment. These are issues that can significantly complicate patient management and often require adjustments in management.

The mucosa of the oral cavity recovers in a manner similar to the skin; however, there is residual compromise of the oral cavity environment, which is long-standing in nature. The floor of the mouth is taut and lacks mucosal redundancy; therefore, it is susceptible to injury and heals less well than other structures in the oral cavity. If ulceration occurs, debris and particles can sequester in the open space and cause necrosis of underlying structures including the mandible. This requires careful management including hyperbaric therapy and surgery. These issues are challenging and may become more frequent as radiation therapy moves to more hypofractionation protocols for head and neck cancer. Patients have dry mouth due to radiation dose to the parotid glands, the submandibular glands, and the often overlooked submucosal gland structures that provide moisture to the mucosal surface. These microscopic glands are ubiquitous in distribution throughout the mucosal surface and create secondary issues for dentition in adults and children. Although teeth do not self-renew and by default are not directly affected by radiation, treatment affects the oral cavity environment and both the growth and development of teeth in children and the oral cavity environment for adults. The saliva becomes more acidic and prone to fungal overgrowth. The gingival mucosa becomes thin and denuded. These changes can lead to chronic decay and demineralization of teeth. Fluoride mouthwash with periodic use of baking soda/water rinses helps deacidify the oral cavity and promotes more optimal oral hygiene; however, these changes are often unrelenting and difficult to control. Optimal radiation planning strategies with intensity modulation may help mitigate these changes moving forward. Building a strong relationship with dental medicine helps facilitate optimal patient care [15, 34].

Motility of the gastrointestinal tract is not a well-described side effect of radiation management; however, this is becoming a more visible issue in the management of head and neck malignancies and esophageal cancers [37]. Contractility of the medial constrictor muscles of the hypopharynx is now described and appears to mimic swallowing issues often seen in patients with neurodegenerative disorders [37]. At times these changes are related to treatment-associated dermal and interstitial edema, and addressing the edema through lymphedema clinics can be very helpful. As more patients survive their primary malignancy, these issues are becoming more frequent and impose significant restrictions on patient recovery from treatment. Delayed gastrointestinal emptying associated with antral fibrosis and denudement of the mucosa of the gastric lining cells can be seen in patients treated to the gastric region, usually at doses of greater than 45 Gy to the gastric region or the gastric resection site.

Late effects to both small and large bowel include every tissue component. There is atrophy of the mucosa resulting in limited absorption of protein, carbohydrate, and fat. This contributes to various degrees of malabsorption syndromes and inconsistent bowel function. If there is previous abdominal surgery, the bowel may be fixed in position resulting in stenosis and ulceration requiring surgery. Relatively little is known about effects to the exocrine and endocrine pancreas although atrophy of the pancreas can be seen on imaging years after treatment of an upper abdominal malignancy [15, 34].

Liver

There is renewed interest in defining radiation dose effects to the liver as stereotactic body radiosurgery techniques have been effective in the treatment of both metastatic disease to the liver and primary hepatocellular carcinoma. Sequela to the liver, known as radiation-induced liver disease (RILD), is driven by the volume of the liver treated and the functional status of the liver at the time of therapy. Both primary and metastatic diseases can impose varying degrees of venoocclusive changes in the parenchyma, and treatment can induce scar tissue that can limit the functional status of the remaining liver. Magnetic resonance imaging has become a valuable tool in validating the degree of veno-occlusive changes and has evolved into a quantitative metric in predicting possible radiation-associated liver injury prior to the administration of radiation therapy [38]. Investigators are using metrics identified on dynamic contrast imaging to establish the appropriate radiation dose to target. The threshold of injury to the liver is significantly decreased when the entire organ volume is treated. Sioshansi et al. have demonstrated injury to the diaphragm without changes in the chest wall resulting in chronic pleuritic pain among patients undergoing stereotactic body radiosurgery [39]. The liver is sensitive to interactions with chemotherapy perhaps best demonstrated in the pediatric population when radiation therapy is delivered with actinomycin D chemotherapy for Wilms' tumor. Sequela can include a dramatic decrease in blood counts as well as changes consistent with liver failure including coagulation disorders [34].

The mechanism associated with therapy-associated hepatic injury is related to direct injury to hepatocytes and

the structural integrity of their organization. The injury is expressed through the associated disorderly repair of hepatocytes coupled with nodular regeneration limiting normal function of the liver. Investigators are currently describing two forms of radiation-induced liver disease. The traditional form of injury (classical) is imposed on pre-injury normal hepatic parenchyma. In these patients, abdominal pain, hepatomegaly, anicteric ascites, and elevation of alkaline phosphatase are seen as the clinical manifestation of injury. In nontraditional (nonclassical) injury, patients often present with jaundice and elevation in transaminase enzymes. Biopsy of the liver will demonstrate edema of the endothelium and narrowing of hepatic venules creating retrograde congestion and veno-occlusive changes. Microthrombi can develop which further exacerbates congestion. Models have demonstrated that an increase in transforming growth factor beta is associated with increasing degree of injury. Hepatic stellate cells are responsible for regeneration of hepatocytes as well as secretion of various growth factors supporting development and regeneration. However, if the reticulum architecture is damaged, the hepatocytes do not have an infrastructure platform to repair in an orderly manner. Disorganized repair is nodule regeneration. Because of a larger separation between hepatic parenchyma and the blood supply, the liver has less meaningful function. This is reflected in the fact that radiation-induced liver disease has a lower threshold for injury when therapy is applied to the liver with pre-existing injury prior to radiation therapy. It is thought that the mean dose threshold for injury in the otherwise healthy liver is 30 Gy while the mean dose threshold for patients with pre-existing injury or lower based on the degree of pre-existing comorbidity.

Because there are no identified therapies to mitigate or treat injury, prevention of injury is essential for current cancer management. Modern radiation therapy techniques including image guidance and motion management have significantly improved patient care and limited the impact of therapy on normal tissue function. Modern imaging techniques including indocyanine retention and magnetic resonance imaging have served as excellent surrogates for liver function which have supported appropriate targeting and radiation target dose and dose fractionation [38, 40–42].

Kidney

The kidney, similar to the liver, is a relatively late-responding radiosensitive critical organ. Radiation doses of greater than 20 Gy in 2 Gy fractions can result in renal damage with anemia and hypertension. Although not yet validated through clinical trials, the threshold for injury is thought to be lower when nephrotoxic chemotherapy is used with radiation therapy. Using intensity modulation and image guidance, radiation oncologists can be creative with partial volume therapy and spare as much renal parenchyma as possible. Nevertheless, in comparison with siblings, there is an increased risk of renal failure in the cancer survivor; therefore, investigators should attempt to limit renal dose to as little as possible during treatment planning. To date this issue has not yet been defined as a point of interest for imaging and avoiding delivery of contrast agents; however, this may become an important issue in the clinic moving forward [15, 34, 36].

The mechanism of injury is multifactorial as radiation therapy likely has impact in all compartments of the kidney including the glomeruli, mesangium, and endothelium. Often injury from therapy is challenging to distinguish from more typical forms of injury including hypertensive renal disease. Oxygen-related species generated as secondary byproducts of therapy are thought to directly impact glomeruli with denuded tubules permitting entry of toxins into interstitial tissues. The development of scar within the kidney and subsequent decrease in size of the affected area are thought to be driven in part by transforming growth factor beta. The sequelae of injury include hypertension which needs to be managed aggressively and followed with rigor. Often the use of angiotensin-converting enzyme inhibitors (ACE inhibitors) and angiotensin receptor blockers is helpful in management. As per other sites of normal tissue injury as noted, the best therapy is prevention. Care and attention to radiation therapy treatment planning with intentional conformal avoidance of renal parenchyma during therapy is essential for the modern cancer patient [43-45].

Lung

As with the liver and kidney, the lung is a very sensitive intermediate to a late-responding tissue. In extreme situations, injury to the therapeutic lung can be life-threatening. There are generally two periods of damage that can be identified. Pneumonitis (period of active inflammation) can occur 2-6 months after completion of radiation therapy, and fibrosis can occur years after treatment delivery. During the pneumonitis phase of injury, there is active inflammation often visible on thoracic imaging. If the patient is asymptomatic, observation is a reasonable approach. Symptoms including cough and shortness of breath associated with these imaged changes are often managed by corticosteroids and antibiotics as appropriate [46]. There are reports of radiation injury to the lung tissue outside of the radiation treatment field. Although felt to be spurious at initial review, investigators have suggested that production of nitric oxide gas as a byproduct of radiation-induced injury may play a role in generating injury in other parts of pulmonary parenchyma not directly in the radiation therapy treatment region [47]. Fibrosis as a late change can result in parenchymal scar as

well as pleural and pericardial effusions resulting in limitation of pulmonary reserve and chronic need for supplemental oxygen. Modern radiation techniques including the use of motion management and intensity modulation may limit the risk of injury by limiting the volume of parenchyma receiving higher doses. Dose-volume histogram analysis also suggests that it is likewise important to limit the volume of normal lung parenchyma receiving 20 Gy. Modern radiation therapy techniques seek to limit the volume of lung receiving both high and low doses. Interactions with other pulmonary toxic agents such as bleomycin play a key role in evaluating the dose-volume effect of radiation therapy. This is especially important in patients treated for Hodgkin lymphoma. Recent studies do reveal an increased risk of chronic pulmonary disease in cancer survivors in comparison with siblings [15, 24, 25, 48].

The mechanism of pulmonary injury is directly related to damage to lung parenchyma. Radiation both primarily generates injury and secondarily creates reactive oxygen and nitrogen species that produce oxidative injury to cell structures leading to cell death. Predominantly, type 1 pneumocytes are directly injured by radiation. Type 2 pneumocytes are less common. Although they can de-differentiate into type 1 pneumocytes, they are conversely stimulated by radiation and exhibit hyperplasia and growth following exposure. Hyperplasia is associated with growth factor production including removal of debris. Inflammatory cells migrate into the region which leads to fibroblast production and fibrosis. Initially the injury is related to cell death and damage to alveolar/capillary structures which ultimately lead to disorganized repair and fibrotic lung tissue.

The degree and severity of injury are often driven in large part by the volume of parenchyma treated. Other variables influencing outcome include baseline pulmonary function and medical comorbidities. Chemotherapy including immunotherapy can have a significant influence on outcome including patients who have favorable metrics with respect to radiation therapy treatment volumes. Radiation metrics including volumes of regions receiving high, moderate (V20, volume receiving 20 Gy), and low (V50, volume receiving 5 Gy) dose each contribute to a normal tissue outcome with varying degrees of influence, often in a patient-specific manner. Further research is needed to see how these variables relate to each other as predictive indicators of function and outcome [49–51].

Case Study

Two patients in Figs. 49.1 and 49.2 represent therapy-driven pneumonitis outlining the radiation therapy field while on immune checkpoint inhibition with improvement after withdrawal of the drug.

T. J. FitzGerald et al.



Fig. 49.1 Case study. Patient 1. Therapy-driven pneumonitis outlining the radiation therapy field while on immune checkpoint inhibition with improvement after withdrawal of the drug. (a) Before immuno-

therapy withdrawal. (b) After immunotherapy withdrawal. (Courtesy of the Department of Radiation Oncologic, University of Massachusetts Medical School, Worcester MA)



Fig. 49.2 Case study. Patient 2. Therapy-driven pneumonitis outlining the radiation therapy field while on immune checkpoint inhibition with improvement after withdrawal of the drug. (a) Before immuno-

therapy withdrawal. (b) After immunotherapy withdrawal. (Courtesy of the Department of Radiation Oncologic, University of Massachusetts Medical School, Worcester MA)

Heart and Peripheral Vessel

Historically the heart was considered to be a late-responding tissue with exception of pericarditis, which could occur during or shortly following radiation therapy, especially in patients with generous cardiac volumes in the treatment field. Patients typically present with sharp, anterior chest pain and pericardial fluid causing shortness of breath and a low-grade fever. Anti-inflammatory medication and treatment interruption alleviate symptoms. With modern cardiology evaluation techniques including magnetic resonance and nuclear medicine studies, we can now identify previously unforeseen cardiac events. This is important as studies are suggesting an association between radiation therapy and the development of cardiovascular disease when the heart is an unintended target of treatment. Tangential irradiation to the left breast as treatment for breast cancer can deliver a measureable mean dose to the heart even with intensity modulation techniques [52].

Anterior-posterior treatment techniques used to treat Hodgkin lymphoma with historical non-image-guided techniques resulted in full-dose radiation therapy to multiple critical cardiac structures. In reviewing the anterior-posterior cardiac anatomy using traditional radiation therapy treatment fields for Hodgkin lymphoma, multiple critical structures reside in the parallax of the vertebral body including the primary cardiac vessels, the electrical conduction nodes, and the aortic valve. The mitral valve resides generally 2 cm lateral to the left edge of the vertebral body in patients without chronic lung disease. Therefore, traditional radiation treatment for Hodgkin lymphoma included many cardiac structures treated at high dose, placing these tissues at risk for the lifetime of the patient. Modern cardiac imaging using multiple platforms reveals segments of the myocardium, which can demonstrate dyskinesis in small segments of the cardiac myocardium (apex) after tangential radiation therapy for breast cancer. The clinical importance of these findings is uncertain. Intensity modulation decreases radiation dose to the heart; however, there is a population treated with radiation therapy in the pre-intensity modulation era that will be at risk for the next several decades for heart disease. Recent studies demonstrate a significant risk of heart disease in the cancer survivor compared to their siblings [31, 39, 48, 52–65].

Large peripheral vessels were historically viewed as resistant to radiation therapy; however, as we begin to move forward with hypofractionation protocols, reports of injury are being reported. During historical times when treatment was delivered with orthovoltage therapy, carotid injury was described recognizing that dose to carotid was likely much higher than the reported tumor dose. Pathologies of injury include intimal hyperplasia and weakening of the carotid muscle. There are reports of fistula formation and sudden death due to rupture of the carotid vessels [31, 62]. Reports of injury to other large vessels (subclavian, femoral, etc.) were reported when there was overlap with radiation therapy treatment fields necessitating large radiation dose to a tubular structure [62]. With modern radiation therapy and traditional fractionation strategies, radiation injury to large vessels is uncommon. However, with higher daily doses to tubular structures, late injury can result and become clinically important. Symptomatic injury to veins is less common, and injury to capillaries can be visible at radiation doses of 50 Gy. This is an area, however, where retreatment of second cancers may predispose to injury in future patients.

The mechanism associated with the impact of radiation therapy on the heart and blood vessels is driven by inflammation related to the trauma of injury coupled with late fibrosis and functional instability. Intimal damage to cardiac vessels followed by cell proliferation in a disorganized manner can lead to premature coronary artery disease and atherosclerosis. Damage to cardiac valves can lead to late fibrosis and calcification which are able to cause both stenosis and insufficiency. Myocardial and pericardial inflammation can lead to muscle dysfunction, cardiomyopathy, and congestive heart failure. Pericarditis can be seen in situations where a large volume of heart receives radiation therapy and acute inflammation can lead to constriction of function. Electrical conduction deficits are seen and thought to be related to fibrosis in conduction pathways. Multiple chemotherapy agents influence cardiac health, and their relationship to injury imposed by radiation therapy remains under evaluation.

To date there is no mitigation management and therapeutic intervention available to manage cardiovascular disease driven by radiation therapy. Surgical teams need to be seasoned and expert in managing these patients. Often these experts need to both review and understand the nature of treatment and how it was applied to optimize outcome for each patient. The most optimal treatment is prevention by a seasoned radiation oncologic team with advanced knowledge of cardiac anatomy and how it can be applied to therapy with conformal avoidance and image guidance with breath-hold techniques to decrease cardiac dose. With modern radiation therapy, therapy targets are designed in four dimensions, and conformal avoidance can be applied as best as possible. Cancers are not convenient, and often cardiac structures receive both high- and low-dose radiation and cannot be avoided. In this cohort of patients, it is best to involve onco-cardiology early in the survivorship process to best ascertain imaging and functional analysis to optimize posttherapy care and identify issues in their early development when intervention may be best utilized and optimized [66].

Central and Peripheral Nervous System

The brain has several categories of cells susceptible to injury including the glia (support cells), primary neurons, and blood vessels. All of these tissues are generally considered as late-responding tissues; therefore, most sequelae occur as late events. The most important sequela is necrosis, which can occur within 6 months of radiation therapy; however, reports of late injury indicate that events can occur several years after treatment. Necrosis is seen more often now that radiosurgery techniques are used more commonly in patient care. Rarely, demyelinating syndromes can occur in the central nervous system. These syndromes can also occur in the spinal cord. Reversible syndromes can occur in the spinal cord in doses as low as 35 Gy; however, irreversible changes including myelitis begin to occur at doses of 45-50 Gy with traditional fractionation and appear to incrementally increase with larger radiation dose and larger volume of the spinal cord included in the treatment field. Toxicity may be increased with the addition of neurotoxic chemotherapy including *cis*-platinum, vinblastine, Ara-C, gemcitabine, and methotrexate. Peripheral nerves can likely tolerate a higher dose of radiation therapy as the cauda equina, and larger nerves appear to tolerate radiation doses in excess of 55 Gy without evidence of injury [67]. Visual field changes are seen in radiation doses higher than 5400 cGy to the optic nerve

and chiasm [68]. It is thought that the chiasm is sensitive to radiation therapy as it has an end-arterial blood supply. This was first described in patients treated to the pituitary gland for pituitary adenomas using daily treatment fractions of greater than 200 cGy/day [69]. Therefore, with modern-day image guidance and partial volume therapy, some investigators believe that tolerance of these structures may be higher than described in the historical literature. The cochlea can be affected by radiation, and this effect can be more pronounced at lower doses with the use of chemotherapy including *cis*-platinum. Historically the lens is very sensitive to radiation therapy with cataract formation identified at very low dose (500 cGy) [9, 70].

Brachial plexopathy has been described in breast cancer patients treated to peripheral lymph nodes. Although the radiation dose threshold for injury is thought to be 5400 cGy, this is an uncommon side effect for patients treated with higher doses of radiation for head and neck cancer. The prevailing thought is that the more sensitive part of the plexus is the region where the nerve bundles coalesce immediately inferior to the lateral third of the clavicle. In the early days of radiation therapy, this area was calculated using an anterior field to a depth of 5 cm. The nerves can be as superficial as 1 cm below the skin surface; therefore, the nerve region received a higher percent dose and in some case would have received the equivalent of a high daily fraction (>120% of the prescribed dose). This is compounded further by the use of posterior axillary boosts, which were often viewed and calculated as separate fields with exit overlap at the egress points of the brachial plexus into the upper extremity. Therefore, it is entirely possible that the threshold dose for brachial plexus injury may be higher than described in historical literature due to unintentional overlap of radiation field volumes and unspecified increases in daily fraction size to a critical target. Today, using three-dimensional volumetrics, the axilla is a volume and modern planning permits more uniform radiation dose distribution through a volume than two-dimensional treatment planning constructs [71].

The mechanism of damage to the central nervous system and peripheral nerves is multifactorial in origin. All forms of irradiation including application of orthovoltage and electrons are associated with injury. Acute changes from therapy generally affect cells with rapid self-renewal potential, relatively few cell systems in the central nervous system divide frequently, and therefore injury is generated through multiple pathways. Although damage is generated to DNA through traditional mechanisms, oxidative species generates injury to lipid bilayers, cell adhesion molecules, blood vessels, and mitochondria. The coupled imprint of primary and subcellular damage alters cell architecture and structure, thus impacting the cellular microenvironment. Vascular permeability is increased which promotes penetration of drugs into the central nervous system. Repair can be generated by stem cells residing in multiple sites of the central nervous system. If the cellular architecture and infrastructure are damaged, repair including remyelination becomes fragmented and disorganized. Irregular and disorganized regeneration of lipid bilayers is seen as leukoencephalopathy. Often injury and tumor recurrence have identical features on MR sequences and can be indistinguishable. Demyelination is frequently seen in peripheral nerves as the primary manifestation of injury as neural stem cells are sparse in peripheral locations.

Injury is often superimposed on previous changes in the central nervous system including vascular and degenerative changes, and it is often difficult to accurately assess the role of therapy on both imaging and neurocognitive testing. Medication interventions are under evaluation at this time to decrease neurocognitive changes associated with therapy. As with most organ systems, the most optimal strategy to prevent injury is to apply radiation therapy to titrated volumes when appropriate and to choose fractionation schemes to optimize repair [72, 73].

Reproductive Organs, Genitalia, and Endocrine

Spermatogonia are among the few cell systems that can die an intermitotic death; therefore, the absolute number of sperm cells markedly decreases with modest doses of radiation. The development period of stem cell to spermatozoa is 75 days; therefore, exposure to radiation can induce damage to mature sperm. In regions of nuclear events such as Chernobyl, there are reports of increased neurocognitive and developmental abnormalities in children born to survivors of these events [6]. Most oncologists offer sperm banking to patients with a known direct exposure to radiation therapy. Indirect exposure with scatter radiation dose will often require the use of birth control for two cycles of sperm development (6 months) to lessen risk of damaged sperm in the ejaculate. Leydig cells secrete testosterone, and their specific function is regulated by pituitary gonadotropins, prolactin, and luteinizing hormone. Treatment to the pituitary gland may impose secondary events to gonadal function. Although Levdig cells are more resistant to radiation exposure, there is an incremental decrease of testosterone in doses exceeding 20 Gy. Various chemotherapy agents including vincristine and mechlorethamine (Mustargen) influence sterility. Oocytes are very sensitive to radiation and, as sperm, die an intermitotic death. Because hormonal secretion is associated with follicular maturation, unlike the testicle, treatment of the ovaries results in more immediate suppression of hormonal function. Female genitalia can demonstrate mucosal atrophy and loss of moisture [15, 34, 74]. Radiation therapy to children for pelvic malignancies including rhabdomyosarcoma can result in significant atrophy and maldevelopment of gonadal organs and pelvic anatomy. Cardiovascular health

as well as other medical comorbidities including problems associated with growth can be significantly influenced by diminution of hormonal function at a young age.

Hypothyroidism is a common sequela associated with both surgery and radiation therapy to the low neck. This is quite prevalent in patients treated for head/neck cancer and Hodgkin lymphoma. Neck dissection and primary surgery also influence the incidence of hypothyroidism in patients treated for head and neck cancer. Pituitary therapy creates panhypopituitary syndrome with a need for replacement therapies as appropriate. This can have significant health issues in multiple endocrine organs. Interestingly, there is little data for adrenal function; however, there are reported cases of adrenal malfunction and decreased cortisol with high-dose radiation therapy. More often, adrenal malfunction occurs in patients as a secondary bystander effect to pituitary therapy [15, 34, 75].

Pediatrics

Treatment of children is unique as every cell system has selfrenewal potential; therefore, unlike adults, sequelae are visible and identified in all tissues due to growth and development. The bone and cartilage are key areas that distinguish adults from children. In general terms, with radiation doses of 20 Gy, growth deficits in the bone may be irreversible as stem cells depopulate [34, 76]. The deficits in bone and cartilage development are more visible with higher radiation doses and younger age. In adults, the threshold dose for bone necrosis may be 55 Gy with traditional fractionation strategies; however, there are interesting reports of the use of advanced technology imaging including MR demonstrating sacral fractures in gynecological patients receiving less than 50 Gy to the bone [76]. Radiosurgery techniques, particularly to targets in close approximation to the chest wall, are reinviting injury to the rib and chest wall that is often nonhealing. Treatment techniques including volume-modulated arc therapy [77] appear to play an essential role in decreasing this risk.

Recent studies have demonstrated that long-term survivors of cancer therapy, including radiation, are frailer than their counterparts and acquire chronic diseases at a higher rate. They may also be susceptible to premature death than their sibling counterparts [48].

Conclusion

Intentional and unintentional radiation exposures have a powerful impact on normal tissue function and can induce both short-term and long-term injuries to all cell systems. In the evaluation of acute phase management, assessing radiation dose and exposure is essential to management strategy. Appropriate support can be given to those at risk for serious acute injury. The effects of radiation treatment and exposure, however, last for the lifetime of the patient and can have implications for all organ systems. A broad understanding of these effects is essential for today's acute health-care provider in the emergency department setting.

References

- Hall EJ, Giaccia AJ. Radiobiology for the radiologist. 7th ed. Philadelphia: Lippincott, Williams and Wilkins; 2012.
- Miller DL, Balter S, Cole PE, Lu HT, Berenstein A, Albert R, et al. Radiation doses in interventional radiology procedures: the RAD-IR study: part II: skin dose. J Vasc Interv Radiol. 2003;14(8):977–90.
- Miller DL, Balter S, Wagner LK, Cardella J, Clark TW, Neithamer CD Jr, et al. Quality improvement guidelines for recording patient radiation dose in the medical record. J Vasc Interv Radiol. 2004;15(5):423–9.
- Shope TB. Radiation-induced skin injuries from fluoroscopy. Radiographics. 1996;16(5):1195–9.
- Donnelly EH, Nemhauser JB, Smith JM, Kazzi ZN, Farfan EB, Chang AS, et al. Acute radiation syndrome: assessment and management. South Med J. 2010;103(6):541–6.
- Turai I, Veress K. Radiation accidents: occurrence, types, consequences, medical management, and lessons learned. Central Eur J Occup Environ Med. 2001;7(1):3–14.
- Baranov A, Gale RP, Guskova A, Piatkin E, Selidovkin G, Muravyova L, et al. Bone marrow transplantation after the Chernobyl nuclear accident. N Engl J Med. 1989;321(4): 205–12.
- Wikipedia. Acute radiation syndrome [Internet]. Wikipedia. The free Encyclopedia. Updated. Sep 2020:25. https://en.wikipedia.org/w/ index.php?title=Acute_radiation_syndrome&oldid=940871364
- Fitzgerald TJ, Jodoin MB, Tillman G, Aronowitz J, Pieters R, Balducci S, et al. Radiation therapy toxicity to the skin. Dermatol Clin. 2008;26(1):161–72, ix.
- Citrin D, Cotrim AP, Hyodo F, Baum BJ, Krishna MC, Mitchell JB. Radioprotectors and mitigators of radiation-induced normal tissue injury. Oncologist. 2010;15(4):360–71.
- 11. Patt HM, Tyree EB, Straube RL, Smith DE. Cysteine protection against X irradiation. Science. 1949;110(2852):213–4.
- Brizel D, Overgaard J. Does amifostine have a role in chemoradiation treatment? Lancet Oncol. 2003;4(6):378–81.
- Brizel DM, Wasserman TH, Henke M, Strnad V, Rudat V, Monnier A, et al. Phase III randomized trial of amifostine as a radioprotector in head and neck cancer. J Clin Oncol. 2000;18(19):3339–45.
- 14. Chitra S, Shyamala Devi CS. Effects of radiation and alphatocopherol on saliva flow rate, amylase activity, total protein and electrolyte levels in oral cavity cancer. Indian J Dent Res. 2008;19(3):213–8.
- Bentzen SM. Preventing or reducing late side effects of radiation therapy: radiobiology meets molecular pathology. Nat Rev Cancer. 2006;6(9):702–13.
- Epperly MW, Bray JA, Krager S, Berry LM, Gooding W, Engelhardt JF, et al. Intratracheal injection of adenovirus containing the human MnSOD transgene protects athymic nude mice from irradiation-induced organizing alveolitis. Int J Radiat Oncol Biol Phys. 1999;43(1):169–81.
- Epperly MW, Defilippi S, Sikora C, Gretton J, Kalend A, Greenberger JS. Intratracheal injection of manganese superoxide dismutase (MnSOD) plasmid/liposomes protects nor-

mal lung but not orthotopic tumors from irradiation. Gene Ther. 2000;7(12):1011-8.

- Farrell CL, Rex KL, Kaufman SA, Dipalma CR, Chen JN, Scully S, et al. Effects of keratinocyte growth factor in the squamous epithelium of the upper aerodigestive tract of normal and irradiated mice. Int J Radiat Biol. 1999;75(5):609–20.
- Soule BP, Hyodo F, Matsumoto K, Simone NL, Cook JA, Krishna MC, et al. Therapeutic and clinical applications of nitroxide compounds. Antioxid Redox Signal. 2007;9(10):1731–43.
- Hyodo F, Matsumoto K, Matsumoto A, Mitchell JB, Krishna MC. Probing the intracellular redox status of tumors with magnetic resonance imaging and redox-sensitive contrast agents. Cancer Res. 2006;66(20):9921–8.
- Guo H, Seixas-Silva JA Jr, Epperly MW, Gretton JE, Shin DM, Bar-Sagi D, et al. Prevention of radiation-induced oral cavity mucositis by plasmid/liposome delivery of the human manganese superoxide dismutase (SOD2) transgene. Radiat Res. 2003;159(3):361–70.
- 22. Stickle RL, Epperly MW, Klein E, Bray JA, Greenberger JS. Prevention of irradiation-induced esophagitis by plasmid/liposome delivery of the human manganese superoxide dismutase transgene. Radiat Oncol Investig. 1999;7(4):204–17.
- Burdelya LG, Krivokrysenko VI, Tallant TC, Strom E, Gleiberman AS, Gupta D, et al. An agonist of toll-like receptor 5 has radioprotective activity in mouse and primate models. Science. 2008;320(5873):226–30.
- Anscher MS, Thrasher B, Rabbani Z, Teicher B, Vujaskovic Z. Antitransforming growth factor-beta antibody 1D11 ameliorates normal tissue damage caused by high-dose radiation. Int J Radiat Oncol Biol Phys. 2006;65(3):876–81.
- 25. Anscher MS, Thrasher B, Zgonjanin L, Rabbani ZN, Corbley MJ, Fu K, et al. Small molecular inhibitor of transforming growth factor-beta protects against development of radiation-induced lung injury. Int J Radiat Oncol Biol Phys. 2008;71(3):829–37.
- 26. Massague J. TGFbeta in cancer. Cell. 2008;134(2):215-30.
- Chin MS, Freniere BB, Bonney CF, Lancerotto L, Saleeby JH, Lo YC, et al. Skin perfusion and oxygenation changes in radiation fibrosis. Plast Reconstr Surg. 2013;131(4):707–16.
- Chin MS, Freniere BB, Lo YC, Saleeby JH, Baker SP, Strom HM, et al. Hyperspectral imaging for early detection of oxygenation and perfusion changes in irradiated skin. J Biomed Opt. 2012;17(2):026010.
- Citrin DE. Radiation modifiers. Hematol Oncol Clin North Am. 2019;33(6):1041–55.
- FitzGerald TJ, Bishop-Jodoin M, Laurie F, Lukez A, O'Loughlin L, Sacher A. Treatment toxicity: radiation. Hematol Oncol Clin North Am. 2019;33(6):1027–39.
- Fajardo L, Berthrong M, Anderson R. Radiation pathology. 1st ed. New York: Oxford University Press; 2001.
- Majno G, Joris I. Cells, tissues, and disease: principles of general pathology. 2nd ed. New York: Oxford University Press; 2004.
- 33. Hoppe BS, Laser B, Kowalski AV, Fontenla SC, Pena-Greenberg E, Yorke ED, et al. Acute skin toxicity following stereotactic body radiation therapy for stage I non-small-cell lung cancer: who's at risk? Int J Radiat Oncol Biol Phys. 2008;72(5):1283–6.
- FitzGerald TJ, Aronowitz J, Giulia Cicchetti M, Fisher G, Kadish S, Lo YC, et al. The effect of radiation therapy on normal tissue function. Hematol Oncol Clin North Am. 2006;20(1):141–63.
- Moulder JE, Cohen EP. Future strategies for mitigation and treatment of chronic radiation-induced normal tissue injury. Semin Radiat Oncol. 2007;17(2):141–8.
- 36. Bentzen SM, Constine LS, Deasy JO, Eisbruch A, Jackson A, Marks LB, et al. Quantitative Analyses of Normal Tissue Effects in the Clinic (QUANTEC): an introduction to the scientific issues. Int J Radiat Oncol Biol Phys. 2010;76(3 Suppl):S3–9.
- McCulloch TM, Jaffe D. Head and neck disorders affecting swallowing. GI Motility. 2006.

- Guha C, Kavanagh BD. Hepatic radiation toxicity: avoidance and amelioration. Semin Radiat Oncol. 2011;21(4):256–63.
- Sioshansi S, Rava PS, Karam AR, Lithgow M, Ding L, Xing W, et al. Diaphragm injury after liver stereotactic body radiation therapy. Pract Radiat Oncol. 2014;4(6):e227–30.
- Chapman TR, Kumarapeli AR, Nyflot MJ, Bowen SR, Yeung RS, Vesselle HJ, et al. Functional imaging of radiation liver injury in a liver metastasis patient: imaging and pathologic correlation. J Gastrointest Oncol. 2015;6(3):E44–7.
- 41. Lee IJ, Seong J, Shim SJ, Han KH. Radiotherapeutic parameters predictive of liver complications induced by liver tumor radiotherapy. Int J Radiat Oncol Biol Phys. 2009;73(1):154–8.
- Pan CC, Kavanagh BD, Dawson LA, Li XA, Das SK, Miften M, et al. Radiation-associated liver injury. Int J Radiat Oncol Biol Phys. 2010;76(3 Suppl):S94–100.
- 43. Dawson LA, Kavanagh BD, Paulino AC, Das SK, Miften M, Li XA, et al. Radiation-associated kidney injury. Int J Radiat Oncol Biol Phys. 2010;76(3 Suppl):S108–15.
- 44. Cohen EP, Robbins ME. Radiation nephropathy. Semin Nephrol. 2003;23(5):486–99.
- 45. Bolling T, Ernst I, Pape H, Martini C, Rube C, Timmermann B, et al. Dose-volume analysis of radiation nephropathy in children: preliminary report of the risk consortium. Int J Radiat Oncol Biol Phys. 2011;80(3):840–4.
- Rosiello RA, Merrill WW. Radiation-induced lung injury. Clin Chest Med. 1990;11(1):65–71.
- 47. Sugihara T, Hattori Y, Yamamoto Y, Qi F, Ichikawa R, Sato A, et al. Preferential impairment of nitric oxide-mediated endotheliumdependent relaxation in human cervical arteries after irradiation. Circulation. 1999;100(6):635–41.
- 48. Armstrong GT, Kawashima T, Leisenring W, Stratton K, Stovall M, Hudson MM, et al. Aging and risk of severe, disabling, life-threatening, and fatal events in the childhood cancer survivor study. J Clin Oncol. 2014;32(12):1218–27.
- Hanania AN, Mainwaring W, Ghebre YT, Hanania NA, Ludwig M. Radiation-induced lung injury: assessment and management. Chest. 2019;156(1):150–62.
- Rajan Radha R, Chandrasekharan G. Pulmonary injury associated with radiation therapy - assessment, complications and therapeutic targets. Biomed Pharmacother. 2017;89:1092–104.
- Giuranno L, Ient J, De Ruysscher D, Vooijs MA. Radiation-Induced Lung Injury (RILI). Front Oncol. 2019;9:877.
- 52. Evans SB, Sioshansi S, Moran MS, Hiatt J, Price LL, Wazer DE. Prevalence of poor cardiac anatomy in carcinoma of the breast treated with whole-breast radiotherapy: reconciling modern cardiac dosimetry with cardiac mortality data. Am J Clin Oncol. 2012;35(6):587–92.
- Lenihan DJ, Cardinale DM. Late cardiac effects of cancer treatment. J Clin Oncol. 2012;30(30):3657–64.
- 54. Marks LB, Yu X, Prosnitz RG, Zhou SM, Hardenbergh PH, Blazing M, et al. The incidence and functional consequences of RT-associated cardiac perfusion defects. Int J Radiat Oncol Biol Phys. 2005;63(1):214–23.
- Yusuf SW, Sami S, Daher IN. Radiation-induced heart disease: a clinical update. Cardiol Res Pract. 2011;2011:317659.
- 56. Ng AK, Bernardo MP, Weller E, Backstrand KH, Silver B, Marcus KC, et al. Long-term survival and competing causes of death in patients with early-stage Hodgkin's disease treated at age 50 or younger. J Clin Oncol. 2002;20(8):2101–8.
- 57. Swerdlow AJ, Higgins CD, Smith P, Cunningham D, Hancock BW, Horwich A, et al. Myocardial infarction mortality risk after treatment for Hodgkin disease: a collaborative British cohort study. J Natl Cancer Inst. 2007;99(3):206–14.
- Hooning MJ, Botma A, Aleman BM, Baaijens MH, Bartelink H, Klijn JG, et al. Long-term risk of cardiovascular disease in 10-year survivors of breast cancer. J Natl Cancer Inst. 2007;99(5):365–75.

- 59. Weintraub NL, Jones WK, Manka D. Understanding radiationinduced vascular disease. J Am Coll Cardiol. 2010;55(12):1237–9.
- Lipshultz SE, Sallan SE. Cardiovascular abnormalities in long-term survivors of childhood malignancy. J Clin Oncol. 1993;11(7):1199–203.
- 61. Gayed IW, Liu HH, Yusuf SW, Komaki R, Wei X, Wang X, et al. The prevalence of myocardial ischemia after concurrent chemoradiation therapy as detected by gated myocardial perfusion imaging in patients with esophageal cancer. J Nucl Med. 2006;47(11):1756–62.
- Lam WW, Leung SF, So NM, Wong KS, Liu KH, Ku PK, et al. Incidence of carotid stenosis in nasopharyngeal carcinoma patients after radiotherapy. Cancer. 2001;92(9):2357–63.
- Adams MJ, Hardenbergh PH, Constine LS, Lipshultz SE. Radiationassociated cardiovascular disease. Crit Rev Oncol Hematol. 2003;45(1):55–75.
- Katz NM, Hall AW, Cerqueira MD. Radiation induced valvulitis with late leaflet rupture. Heart. 2001;86(6):E20.
- 65. Slama MS, Le Guludec D, Sebag C, Leenhardt AR, Davy JM, Pellerin DE, et al. Complete atrioventricular block following mediastinal irradiation: a report of six cases. Pacing Clin Electrophysiol. 1991;14(7):1112–8.
- Moslehi J. The cardiovascular perils of cancer survivorship. N Engl J Med. 2013;368(11):1055–6.
- 67. Pieters RS, Niemierko A, Fullerton BC, Munzenrider JE. Cauda equina tolerance to high-dose fractionated irradiation. Int J Radiat Oncol Biol Phys. 2006;64(1):251–7.
- Giese W, Kinsella T. Radiation injury to peripheral and cranial nerves. In: Gutin PH, Leibel SA, Sheline GE, editors. Radiation injury to the nervous system. New York: Raven; 1991. p. 383–403.

- Harris JR, Levene MB. Visual complications following irradiation for pituitary adenomas and craniopharyngiomas. Radiology. 1976;120(1):167–71.
- Stoll BA, Andrews JT. Radiation-induced peripheral neuropathy. Br Med J. 1966;1(5491):834–7.
- Powell S, Cooke J, Parsons C. Radiation-induced brachial plexus injury: follow-up of two different fractionation schedules. Radiother Oncol. 1990;18(3):213–20.
- Greene-Schloesser D, Moore E, Robbins ME. Molecular pathways: radiation-induced cognitive impairment. Clin Cancer Res. 2013;19(9):2294–300.
- Ahles TA, Root JC, Ryan EL. Cancer- and cancer treatmentassociated cognitive change: an update on the state of the science. J Clin Oncol. 2012;30(30):3675–86.
- 74. World Nuclear Association. Nuclear radiation and health effect. London: World Nuclear Association; 2018 [updated June 2018; cited 25 Feb, 2020]. Available from: http://www.world-nuclear. org/information-library/safety-and-security/radiation-and-health/ nuclear-radiation-and-health-effects.aspx.
- 75. Niazi AK, Niazi SK. Endocrine effects of Fukushima: radiation-induced endocrinopathy. Indian J Endocrinol Metab. 2011;15(2):91–5.
- Hopewell JW. Radiation-therapy effects on bone density. Med Pediatr Oncol. 2003;41(3):208–11.
- 77. Ding L, Lo YC, Kadish S, Goff D, Pieters RS, Graeber G, et al. Volume Modulated Arc Therapy (VMAT) for pulmonary Stereotactic Body Radiotherapy (SBRT) in patients with lesions in close approximation to the chest wall. Front Oncol. 2013;3:12.

Mucositis

Jennifer A. Novak and Arya Amini

Introduction

Mucositis is a known potential toxicity associated with several systemic chemotherapies [1], targeted therapies [2], immunotherapies [3], radiation treatments [4], and hematopoietic stem cell transplants (HSCT) [5]. Mucositis is subdivided into two main categories: oral mucositis and gastrointestinal mucositis.

Oral mucositis refers to inflammation and ulcerative injury within the oral cavity. Oral mucositis is one of the most common toxicities among patients receiving chemotherapy or targeted therapy for a variety of cancers. Oral mucositis is also commonly caused by radiation therapy or chemoradiation therapy for cancers involving the head and neck. Additionally, oral mucositis is a known adverse effect of myeloablative therapies preceding HSCT. The most common symptoms of oral mucositis are burning discomfort and pain, which often result in a difficulty eating, drinking, swallowing, and talking. These symptoms lead to decreased oral intake and, in some cases, the inability to meet nutritional and hydration needs. Patients with oral mucositis often have difficulty maintaining good oral hygiene, thereby increasing the risk of severe infection from an odontogenic source.

Gastrointestinal mucositis can occur anywhere along the gastrointestinal tract distal to the oral cavity. Injury to the gastrointestinal mucosa results in ulcerative changes similar to those seen in cases of oral mucositis. However, differences in anatomy, microbial environment, and physiology underlying the development of gastrointestinal mucositis are key reasons for this subclassification. Gastrointestinal mucositis may result in any of the following symptoms, depending on anatomic location: nausea, vomiting, abdominal cramping, diarrhea, rectal pain, or rectal bleeding. Decreased oral intake is a common downstream effect of gastrointestinal mucositis.

J. A. Novak \cdot A. Amini (\boxtimes)

Department of Radiation Oncology, City of Hope National Medical Center, Duarte, CA, USA e-mail: aamini@coh.org Mucositis is a dose-limiting toxicity and may result in premature discontinuation of anti-neoplastic therapies. Further, severe cases of mucositis may require hospitalization. Treatment interruptions due to premature discontinuation of therapy or hospitalization are linked to poor oncologic outcomes such as increased risk of disease recurrence and decreased overall survival.

Case Study

A 65-year-old male with a 35-pack-year smoking history presented to his dentist with a nonhealing oral ulcer. Physical exam revealed a large ulcerative lesion over the left tongue with associated induration of the surrounding tissue. A biopsy of the lesion demonstrated P16-negative squamous cell carcinoma. A CT of the head and neck revealed asymmetric thickening of the left tongue as well as an enlarged left level 2 lymph node. An FDG-PET CT demonstrated FDG avidity associated with the left tongue as well as the left level 2 lymph node, but no evidence of distant metastasis. The patient underwent a hemiglossectomy and bilateral neck dissection demonstrating SCC measuring 3.0 cm in the left tongue with a depth of invasion of 10 mm and multiple positive lymph nodes in the left neck. Margins were focally positive. The patient was recommended to undergo postoperative chemoradiation treatment to the oral cavity and bilateral neck to a maximum dose of 66 Gy in 33 fractions with concurrent weekly cisplatin. During the third week of chemoradiation, the patient complained of discomfort in his oral cavity when eating. Physical exam revealed grade 1 oral mucositis. Over the following 2 weeks, the patient experienced progressive pain in his oral cavity, decreased oral intake, and weight loss exceeding 10% of his total body weight. Physical exam during the fifth week of chemoradiation revealed grade 3 oral mucositis. During the patient's sixth week of treatment, the patient was hospitalized for sepsis and failure to thrive, requiring discontinuation of chemoradiation.





History/Background

Over the past several decades, there have been a few improvements to anti-mucositis agents [6]. While some agents have been shown to decrease the severity and duration of mucositis, there are no known curative agents. Despite the number of randomized studies suggesting a benefit to anti-mucositis agents, the results often lack reproducibility, making development of treatment guidelines a challenge. Thus, mucositis remains a significant problem both for cancer patients and their physicians.

Anatomy

The oral and gastrointestinal mucosae arise from the endoderm and consist of a continuous membrane lining the entirety of the alimentary tract from the oral cavity through the anal canal. There are several layers that comprise the mucosa including most superficially (interiorly) the epithelium, followed by a layer of connective tissue, followed by a muscular layer (muscularis mucosa). The mucosae overlie a layer of dense connective tissue known as the submucosa. The purpose of the mucous membrane is to secrete mucous, providing a barrier to pathogens.

Pathophysiology

The pathogenesis of injury to the oral and gastrointestinal mucosa is characterized by five key phases: initiation, upregulation, signal amplification, ulceration, and healing [7].

Anti-neoplastic therapies, such as chemotherapy and radiation, are known to cause both direct and indirect injuries to the oral and gastrointestinal mucosa. Direct tissue injury occurs when double-stranded DNA breaks induce apoptotic signaling and the production of reactive oxidative species. In turn, pro-inflammatory cytokines are released, such as TNF- α and IL-1 β , and NF- κ B signaling pathways are activated, which play a key role in the pathogenesis of mucositis [8, 9] (Fig. 50.1). Indirect tissue injury occurs when antineoplastic therapies reduce the mucosal barrier through villus blunting and crypt ablation. The combination of direct and indirect tissue injury leads to signal amplification and exacerbated tissue injury.

Several studies have demonstrated the impact of the local tissue microbiome on the pathogenesis of mucositis. Differences in the microbiome in the oral cavity compared to that of the gastrointestinal tract help to explain some of the site-specific reactions to therapeutic agents. Bacterial ligands of resident gastrointestinal flora not only regulate the mucosal immune response to anti-neoplastic agents but also play an important role in mucosal regeneration [10].

The physiologic mechanisms underlying the symptoms caused by mucositis also vary depending on anatomic location. In the small bowel, mucosal injury to enteric cells results in the release of 5-hydroxytryptamine (5-HT), serotonin, and substance P. These chemicals act as neurotransmitters and stimulate the dorsal brainstem, leading to symptoms such as nausea and vomiting [6, 11]. In the colon, mucosal injury results in the release of pro-inflammatory cytokines, which stimulate a neuronal feedback signal resulting in diarrhea.



Fig. 50.1 The pathogenesis of mucosal injury. Anti-neoplastic therapies (1) cause both direct (2) and indirect (3) mucosal injuries, which result in activation of TNF- α and IL-1 β , and NF- κ B signaling pathways

(4). The local tissue microbiome exerts immunomodulatory effects and plays a role both in mucosal damage and regeneration (5). (Adapted from Bowen et al. with permission [8] Springer Nature)

Epidemiology

Approximately one fifth of advanced cancer patients will experience mucositis [12]. Of those, 60% will experience a decrease in oral intake, 40% will experience severe impairment of nutrition or hydration, and 6% will experience complete prevention of nutrition or hydration due to mucositis. Of the approximate 440,000 patients receiving chemotherapy annually in the USA, more than one third will experience mucositis [13]. Of the 20,000 hematopoietic stem cell transplants in the USA annually, at least 87% of patients will experience mucositis [14].

Patients with cancer arising from the head or neck are at particularly high risk of developing oral mucositis, with an odds ratio of 6.31 [6]. In 2020, there will be an estimated 65,630 new cases of head and neck cancer diagnosed in the USA [15] and 650,000 new cases worldwide [16], and at least 80% undergoing radiation for head and neck cancer will experience mucositis [17].

Individuals at the greatest risk for the development of mucositis include the pediatric and geriatric populations. 80% of pediatric patients receiving chemotherapy experience oral mucositis [18] and 99% of pediatric patients undergoing hematopoietic stem cell transplant [19] experience either oral or gastrointestinal mucositis. Geriatric patients comprise a majority of new cancer cases and tend to have poor tolerance of oncologic therapies. Further, geriatric patients often recover more slowly from anti-neoplastic toxicities than do younger patients.

Health Economics

Cases of oral and gastrointestinal mucositis have a significant economic impact. A study published in 2008 reported that the total median medical costs of patients with mucositis or pharyngitis related to anti-neoplastic therapy were \$39,313 [20]. These costs were approximately double that of the total median medical costs of \$20,798 for patients without therapy-induced mucositis. The largest contributor to increased medical costs for patients with mucositis is extended inpatient hospitalizations, which can occur as a result of the potential sequelae of mucositis, such as infection or nutritional deficiency. A report based on data from 2012 estimated a combined cost of \$15,500 for every inpatient hospitalization due to severe mucositis [21].

Toxic mucositis results in annual costs of \$13.23 billion in the USA [6]. Cases of grades 1-2 mucositis account for roughly 60% of the total cost burden, and cases of grades 3-4 mucositis account for the remaining 40%.

Indirect costs of mucositis relate to the amount of time spent by physicians and clinical staff managing patient's symptoms of mucositis. Nurses and physicians spend an additional 9.0 and 5.7 hours, respectively, per head/neck cancer patient managing mucositis during their treatment course [22].

Presentation

The symptoms associated with mucositis vary depending on the anatomic location of the mucosal injury. Patients with oral mucositis initially present with a burning sensation in the oral cavity or oropharynx, which progresses to pain. Pain associated with oral mucositis can range from mild to severe and may result in difficulty chewing or swallowing, unintentional weight loss, or the inability to maintain oral hygiene. Oral mucositis increases the risk of infection and, in some cases, can result in hospitalization or premature discontinuation of anti-neoplastic therapy. Based on patientreported data, the approximate duration of oral mucositis in patients undergoing radiation treatment is 70–84 days and 68–102 days in those undergoing chemotherapy (4-6 cycles) [17, 23].

Patients with gastrointestinal mucositis present with inflammatory symptoms and, depending on the anatomic location of mucosal injury, can be characterized as pharyngitis, esophagitis, gastritis, colitis, or proctitis. Patients with mucositis-related pharyngitis or esophagitis may experience odynophagia, dysphagia, or retrosternal pain. Patients with mucositis-related gastritis or colitis may experience nausea, vomiting, abdominal cramping, or diarrhea. Patients experiencing mucositis-related proctitis may experience pain with bowel movements or rectal bleeding.

Diagnosis

Oral mucositis can be diagnosed by inspection of the oral cavity for erythema, edema, and ulceration of the oral mucosa. Detection of mucosal injury distal to the oral cavity is more challenging. Endoscopy allows for the visualization of mucosal changes; however, endoscopy is not recommended in patients with suspected or presumed mucositis, as it may lead to further tissue injury. There are ongoing studies of non-invasive methods to aid in the diagnosis of mucositis, though these methods have not been adopted as common practice. For example, esophageal mucositis is associated with mucosal thickening identified on CT-based imaging [24]. Similarly, bowel wall thickening visualized on ultrasound may indicate inflammation related to gastrointestinal mucositis [25]. Additional studies are aimed at identifying one or more biomarkers associated with mucositis. At this time, gastrointestinal mucositis largely remains a clinical diagnosis made based on the presence of patient-reported symptoms.

Table 50.1 Comparison of mucositis assessment tools

		G 1 2	G 1 2	G 1 4	0 1 5
	Grade I	Grade 2	Grade 3	Grade 4	Grade 5
RTOG/ EORTC [27]	Irritation or erythema of the mucosa; may experience mild pain; may require topical analgesic; may require soft diet; <5% weight loss from pretreatment baseline; nausea, abdominal discomfort, and increased frequency of bowel habits not requiring intervention	Patchy mucositis that may produce an inflammatory serosanguinous discharge; may experience moderate pain; may require narcotic analgesics; may require puree or liquid diet; <15% weight loss from pretreatment baseline; may experience nausea, vomiting, abdominal pain, or diarrhea requiring intervention	Confluent fibrinous mucositis; may include severe pain requiring narcotic; >15% weight loss from pretreatment baseline requiring feeding tube, IV fluids or hyperalimentation; nausea, vomiting, abdominal pain/ distention despite medication	Ulceration, hemorrhage, or necrosis; perforation or fistula; ileus; obstruction; GI bleeding requiring transfusion; abdominal pain requiring tube decompression or bowel diversion	N/A
WHO [26]	Mild symptoms; oral soreness, erythema	Moderate symptoms; oral erythema or ulcers, solid diet tolerated	Severe symptoms; oral ulcers with extensive erythema, requiring liquid diet only	Life-threatening symptoms; requirement for parenteral or enteral support	N/A
NCI CTCAE [28]	Erythema of the mucosa; minimal discomfort; normal diet; not interfering with function; intervention not indicated	Patchy ulcerations or pseudomembranes; symptomatic but can eat and swallow; modified diet; no interference with activities of daily living; medical intervention indicated	Confluent ulcerations or pseudomembranes; bleeding with minor trauma; symptomatic and unable to adequately aliment or hydrate orally; stool incontinence or other symptoms interfering with activities of daily living	Tissue necrosis; significant spontaneous bleeding; life- threatening consequences	Death

RTOG Radiation Therapy Oncologic Group, EORTC European Organisation for Research and Treatment of Cancer, WHO World Health Organization, NCI CTCAE National Cancer Institute Common Terminology Criteria for Adverse Events

There are several criteria by which the severity of mucositis can be classified. Grading systems include the World Health Organization (WHO) scale for mucositis [26], the Radiation Therapy Oncologic Group (RTOG) and European Organisation for Research and Treatment of Cancer (EORTC) toxicity criteria [27], and the National Cancer Institute Common Terminology Criteria for Adverse Events (NCI CTCAE) [28] (see Table 50.1). Examples of oral mucositis grades 1–4 based on the WHO grading scale can be seen in Fig. 50.2. Other assessment tools include the Oral Mucositis Assessment Scale (OMAS) [29] and the patient-reported outcome measure (PROM) scale [30].

Prophylaxis and Management of Oral Mucositis

Anti-inflammatory Agents

Benzydamine is an anti-inflammatory agent used for the prevention and treatment of oral mucositis. Benzydamine inhibits prostaglandin synthetase, thereby blocking the production of pro-inflammatory cytokines such as TNF- α and IL-1 β . Benzydamine is currently the only agent for which there is level 1 evidence supporting its use in the management of mucositis. Benzydamine mouthwash can be used both for the prophylaxis and treatment of oral mucositis in patients receiving chemotherapy alone, radiation treatment alone, or concurrent chemoradiation to the head/neck [31, 32]. Randomized studies have demonstrated that benzydamine improves pain [33] as well as severity of oral mucositis [34]. However, benzydamine may not control radiation-induced oral mucositis at radiation doses greater than 50 Gy [35–37].

Irsogladine maleate is an anti-inflammatory agent that inhibits the production of reactive oxidative species. There are data demonstrating a benefit of systemic irsogladine maleate for the prevention of chemotherapy-induced oral mucositis [38].

Rebamipide is an anti-inflammatory agent with several mechanisms of action that ultimately promote mucosal protection. One randomized study demonstrated a decrease in the severity of oral mucositis in patients receiving chemoradiation treatment in patients using rebamipide mouthwash compared to placebo [39].

Oral Cryotherapy

Oral cryotherapy is the method of cooling the oral cavity either by ice or cold liquids. Oral cryotherapy results in local vasoconstriction, reducing the exposure of mucous membranes in the oral cavity to cytotoxic agents present in systemic therapies. Evidence supports its use in the prevention of chemotherapy-induced mucositis, especially in patients receiving fluorouracil-based chemotherapy or highdose melphalan-based therapy preceding HSCT [40, 41].



Fig. 50.2 Examples of oral mucositis on physical exam according to the WHO [26]. Panel (**a**) is an example of WHO grade 1 mucositis, characterized by erythema of mucous membranes. Panel (**b**) is an example of WHO grade 2 mucositis, characterized by the presence of

oral ulcers. Panel (c) is an example of WHO grade 3 mucositis, characterized by the presence of extensive ulcers causing a need for liquid diet. Panel (d) is an example of WHO grade 4 mucositis, characterized by the presence of severe mouth sores inhibiting oral intake

Ease of administration and cost-effectiveness are additional benefits of oral cryotherapy. In comparing anti-mucositis agents, oral cryotherapy is suggested to be the best intervention to reduce the incidence of oral mucositis with a few side effects [42].

Palifermin

Palifermin is a keratinocyte growth factor that induces the epithelial cell proliferation in the oral and gastrointestinal mucosa. Palifermin results in cytoprotective effects, decreases the production of pro-inflammatory cytokines, and promotes mucosal regeneration. Palifermin is recommended for the prevention of oral mucositis in patients undergoing high-dose chemotherapy and total body irradiation prior to HSCT. Data have suggested that palifermin reduces both the severity and duration of oral mucositis in patients undergoing HSCT [43]. The recommended dose of palifermin for the prevention of oral mucositis is 60 ug/day for 3 days prior to conditioning treatment and 3 days following stem cell transplant [5, 44]. A common side effect of palifermin is taste disturbance [42].

Photobiomodulation

Photobiomodulation (PBM), also referred to as low-level light therapy or low-level laser therapy, consists of nonionizing light sources. PBM has been shown to promote wound healing and decrease inflammation following mucosal injury. Several randomized studies have demonstrated the benefit of intraoral PBM for the prevention of oral mucositis in patients undergoing HSCT [45]. Studies have also demonstrated a benefit of intraoral PBM for the prevention of radiation-induced oral mucositis. PBM used for the prevention of oral mucositis should be delivered at a wavelength of 650 nm, power of 40 mW, and tissue energy dose of 2 J/cm² [5, 44].

Opioid Analgesics

Patient-controlled morphine has been recommended for the management of severe oral mucositis in patients undergoing HSCT [5]. There is no evidence to demonstrate the superiority of continuous infusion morphine over patient-controlled morphine; however, patient-controlled morphine is associated with less total consumption [46]. Topical morphine, in the form of mouthwash, has been shown to reduce the severity of oral mucositis in patients undergoing radiation, chemotherapy, or chemoradiation therapy for cancer of the head/neck [47, 48]. Transdermal fentanyl has been shown to be both safe and effective in managing pain due to chemoradiotherapy-related oral mucositis [49].

Gabapentin

Gabapentin is a structural analog of gamma-aminobutyric acid (GABA), an amino acid that acts as an inhibitory neurotransmitter. Gabapentin is known for its effectiveness as an antiepileptic, though it is also commonly used in the treatment of neuropathic pain. There are data supporting the use of gabapentin for reduction of pain associated with radiotherapy-induced oral mucositis [50, 51]. Gabapentin can reduce or completely eliminate the need for opioid pain medications. The recommended dose of gabapentin for oral mucositis is 2700 mg daily.

Doxepin

Doxepin hydrochloride is a tricyclic antidepressant shown to have analgesic and anesthetic effects when used topically. Studies have demonstrated a statistically significant reduction in pain related to oral mucositis among patients undergoing radiotherapy for head and neck cancer using doxepin oral rinse [52, 53].

Nutritional Supplements

Glutamine is an amino acid that promotes cellular proliferation and survival. Oral glutamine has been shown to delay the onset, and reduce the severity of, oral mucositis in patients receiving chemoradiotherapy to the head and neck [54, 55]. The recommended dosage is 10–30 mg per day, and it is available in various formulations including a mouth rinse as well as oral forms. Zinc is an electrolyte that promotes homeostasis, wound healing, and immune responses. Several randomized studies have demonstrated the benefit of zinc in the prevention of oral mucositis among patients receiving radiation or chemoradiation treatment to the head/neck [56, 57].

Elemental diet formulations are nutrient-rich liquids and may be beneficial in preventing oral mucositis in patients undergoing radiation treatment to the head/neck [58].

Vitamin E is an antioxidant with anti-inflammatory properties. Studies have shown a benefit of vitamin E for the treatment of chemotherapy-induced oral mucositis [59, 60]. Topical vitamin E, available in a mouth rinse, appears to be more effective than in pill form [61]. There are data suggesting a benefit to vitamin E in preventing radiation-induced oral mucositis as well [62].

Natural Agents

Honey has been shown to reduce the severity of radiationinduced oral mucositis [63]. Honey functions both as an antimicrobial and also to maintain the integrity of the mucosal epithelium. The high sugar content of honey increases the risk of dental caries; therefore good dental hygiene practices are crucial for patients using honey.

Aloe vera has both anti-inflammatory and antimicrobial properties, and it has been shown to modulate the immune system when delivered orally [64]. There are data to suggest that aloe vera reduces both the incidence and the severity of oral mucositis in patients undergoing radiation to the head/ neck [65, 66]. Additional studies are needed to identify the optimal dose of aloe vera, as high concentrations can result in diarrhea and drug interactions [67].

Curcumin is a compound found in turmeric with antiinflammatory, antioxidant, and antitumor properties [68]. Turmeric mouthwash has been shown to delay the onset of mucositis and decrease the severity of mucositis in patients undergoing radiation therapy for cancer of the head/neck [69, 70].

Oral Care

Professional oral care and routine oral hygiene practices performed at home by patients may reduce the severity of oral mucositis and the risk of infection from odontogenic sources. Studies have shown that professional dental evaluation and treatment reduces the severity of oral mucositis and improves pain associated with oral mucositis [71]. Patient education is recommended to improve patient compliance with oral hygiene practices at home. Patients should also be instructed to drink plenty of water and avoid painful stimuli such as hot or spicy foods or liquids, smoking, alcohol, and objects that may cause trauma to the oral cavity (e.g., ill-fitting dentures). There have been studies demonstrating a benefit to saline or sodium bicarbonate mouth rinses in reducing the severity of oral mucositis. A multi-agent regimen is recommended to optimize oral care in patients undergoing chemotherapy, HSCT, or radiation to the head/neck.

Prophylaxis and Management of Gastrointestinal Mucositis

Amifostine

Amifostine is a prodrug with several mechanisms of action related to the scavenging of free radicals and DNA repair. A recommended dose of 340 mg/m² of intravenous amifostine can be administered for prevention of radiationinduced proctitis [29]. There are also data suggesting a benefit for amifostine to reduce esophageal mucositis in patients undergoing radiation treatment for non-small cell lung cancer [72].

Octreotide

Octreotide is a somatostatin analog that causes arterial vasoconstriction. Octreotide is best known for its use in the management of acute gastrointestinal bleeding. There are data to suggest the use of intramuscular octreotide in the treatment of diarrhea associated with HSCT, but only following a failed trial of loperamide [44].

Sucralfate

Sucralfate is an agent best known for its use in peptic ulcer disease. Sucralfate functions both as an acid buffer and a protective barrier for gastrointestinal mucosa. Prior studies have suggested its benefit for radiation-induced proctitis [73, 74].

Antimicrobials

The use of oral antibiotics ciprofloxacin and metronidazole was studied in a randomized trial for the treatment of chronic radiation-related proctitis and demonstrated a benefit [75, 76].

Antihistamines

Histamine receptor antagonists, such as famotidine, may be useful for the prevention of gastrointestinal mucositis in patients undergoing radiation treatment for prostate cancer, though additional studies are needed [77].

Nutritional Supplements and Dietary Modifications

Numerous modified diets have been studied for efficacy in the prevention and management of gastrointestinal mucositis. There is no clear benefit of fat-modified diets, fiber supplementation, or prebiotic supplementation. However, there are data to support the use of oral glutamine to reduce the severity of radiation-induced esophagitis [78–80] and to decrease the duration of diarrhea in patients undergoing HSCT [81]. Probiotics, or oral formulations of live bacterial species found in normal gut flora, may be useful in the prevention of chemotherapy or radiation-induced diarrhea [82, 83].

Formalin

Topical formalin reduces superficial mucosal blood flow [84], and data suggest a benefit in cases of hemorrhagic radiation-related proctitis [85].

Palifermin

In addition to using palifermin for the prevention of oral mucositis, it has been shown to decrease the severity of chemotherapy-induced and HSCT-related gastrointestinal toxicity [86, 87].

Hyperbaric Oxygen Therapy

Hyperbaric oxygen therapy (HBO) is a treatment modality by which 100% oxygen is delivered in a pressurized chamber. HBO results in the production of reactive oxygen and nitrogen species, which promote tissue healing via several molecular pathways. HBO has been studied in the context of late radiation toxicity, and data support the use of HBO for treatment of late radiation-related proctitis [88].

Disposition/Follow-Up

Patients with a history of therapy-induced mucositis should receive regular oncologic follow-up, at which time they can be evaluated for long-term effects of mucosal injury. Patients recovering from mucositis may benefit from follow-up with a dietician. There are several screening tools to evaluate patients for chronic nutritional deficiency, such as the malnutrition universal screening tool (MUST) and the NUTRISCORE tool [89, 90].

Prognosis/Treatment Complications

Most cases of acute therapy-related mucositis will resolve with time. Significant risks associated with mucositis include severe infections, dehydration, and weight loss. Such infections and severe malnutrition result in unexpected treatment breaks or premature discontinuation of anti-neoplastic therapies and are associated with increased rates of recurrence and worse survival. Patients should be monitored closely with additional nutritional support including a feeding tube if needed and routine IV hydration.

Common Pitfalls

Patients at high risk of oral and gastrointestinal mucositis should be identified prior to initiating therapy. Education regarding the symptoms of mucositis and the importance of oral hygiene and nutrition should precede initiation of therapy. Patients at risk for developing mucositis should undergo thorough evaluation by both a dental health professional and a dietician. Throughout the course of therapy, patients should be screened weekly for weight loss and for symptoms of mucositis. Early intervention with intravenous fluid hydration and/or enteral feeding tube placement is critical in patients experiencing rapid, continuous weight loss and can reduce the risk of hospitalization or treatment break.

Future Needs/Vision

Despite the growth of literature pertaining to anti-mucositis agents, guidelines continue to lack specific recommendations for many agents studied in the prevention and management of oral and gastrointestinal mucositis. The range of anatomic sites of mucositis development, as well as the numerous therapies known to cause mucositis, results in highly specific studies. Furthermore, the distinction between prevention of mucositis and treatment of mucositis is important due to differences in the mechanisms of early compared to chronic mucosal injury. Ongoing studies should further elucidate the pathogenesis of mucositis, and their results may be valuable in identifying potential targets for antimucositis agents.

Residency and fellowship training should include formal didactics pertaining to the diagnosis, documentation, and management of oral and gastrointestinal mucositis.

Health Services/Resource Use

Educational resources for patients and physicians can be found via the Multinational Association of Supportive Care in Cancer (MASCC) [5], an organization that has published its recommendations for the management of mucositis. The European Society for Medical Oncologic (ESMO) and the National Comprehensive Cancer Network (NCCN) have published their own sets of clinical guidelines [44, 91].

Documentation

Mucositis is often underreported by physicians. Patientreported duration and severity of mucositis are typically longer and worse compared to that reported by physicians [92]. The cause of underreporting is multifactorial. However, the attitude among health professionals that toxicities related to therapy are "normal" and fear among physicians that adverse events will reflect poorly on their practice may be contributing factors [93, 94].

Key Points/Pearls

- Pathogenesis of mucositis is mediated by pro-inflammatory cytokines released as a result of both direct and indirect mucosal injuries. Ongoing studies are aimed at elucidating the role of the microbiome in the pathogenesis of mucositis, as well as the process of mucosal regeneration/healing.
- Diagnosis of oral mucositis is based upon physical exam; however, the diagnosis of gastrointestinal mucositis is based on patient symptoms aided by the use of patientreported assessment tools.
- Early nutritional intervention and counseling are recommended for patients at risk of developing mucositis. Nutritional counseling by a dietician specializing in oncologic, dental evaluation by a dental health professional, and education regarding the importance of routine oral hygiene practices are recommended for those patients identified as high risk.
- Evaluation of patients early in the course of therapy for symptoms of mucositis is critical to reduce the likelihood of hospitalization or treatment interruptions, which are associated with worse oncologic outcomes.

References

- 1. Chaveli-López B. Oral toxicity produced by chemotherapy: a systematic review. J Clin Exp Dent. 2014;6(1):e81.
- Arena C, Troiano G, De Lillo A, Testa NF, Lo Muzio L. Stomatitis and VEGFR-tyrosine kinase inhibitors (VR-TKIs): a review of current literature in 4369 patients. Biomed Res Int. 2018;2018:5035217.

- Amy DPB, Shalabi A, Finfter O, Birenzweig Y, Zadik Y. Severe chronic nonlichenoid oral mucositis in pembrolizumab-treated patients: new cases and a review of the literature. Immunotherapy. 2020;12(11):777–84.
- Maria OM, Eliopoulos N, Muanza T. Radiation-induced oral mucositis. Front Oncol. 2017;7:89.
- Lalla RV, Bowen J, Barasch A, Elting L, Epstein J, Keefe DM, et al. Mucositis Guidelines Leadership Group of the Multinational Association of Supportive Care in Cancer and International Society of Oral Oncology (MASCC/ISOO). MASCC/ISOO clinical practice guidelines for the management of mucositis secondary to cancer therapy. Cancer. 2014;120(10):1453–61.
- McCullough RW. US oncology-wide incidence, duration, costs and deaths from chemoradiation mucositis and antimucositis therapy benefits. Future Oncol. 2017;13(30):2823–52.
- Sonis ST. Pathobiology of oral mucositis: novel insights and opportunities. J Support Oncol. 2007;5(9 Suppl 4):3–11. PMID: 18046993.
- Bowen J, Al-Dasooqi N, Bossi P, Wardill H, Van Sebille Y, Al-Azri A, et al. The pathogenesis of mucositis: updated perspectives and emerging targets. Support Care Cancer. 2019;27(10):4023–33. https://doi.org/10.1007/s00520-019-04893-z.
- Kanarek N, Grivennikov SI, Leshets M, Lasry A, Alkalay I, Horwitz E, et al. Critical role for IL-1β in DNA damageinduced mucositis. Proc Natl Acad Sci U S A. 2014;111(6):E702–11.
- Cario E. Toll-like receptors in the pathogenesis of chemotherapyinduced gastrointestinal toxicity. Curr Opin Support Palliat Care. 2016;10(2):157–64.
- Thorpe D, Stringer A, Butler R. Chemotherapy-induced mucositis: the role of mucin secretion and regulation, and the enteric nervous system. Neurotoxicology. 2013;38:101–5.
- Mercadante S, Aielli F, Adile C, Ferrera P, Valle A, Fusco F, Caruselli A, Cartoni C, Massimo P, Masedu F, Valenti M, Porzio G. Prevalence of oral mucositis, dry mouth, and dysphagia in advanced cancer patients. Support Care Cancer. 2015;23(11):3249–55.
- Elting LS, Cooksley C, Chambers M, Cantor SB, Manzullo E, Rubenstein EB. The burdens of cancer therapy: clinical and economic outcomes of chemotherapy-induced mucositis. Cancer. 2003;98(7):1531–9.
- 14. Sonis S, Vera-Llonch M, Ford C, Lu J, Khazanov I, Oster G. Duration of ulcerative mucositis (OM) and outcomes of allogeneic (AL) hematopoietic stem cell transplantation (HSCT) in patients with hematologic malignancies. Biol Blood Marrow Transplant. 2006;12(2 Suppl 1):23 (abstract). https://www.bbmt.org/article/S1083-8791(05)00874-8/fulltext.
- Cancer.Net. Head and neck cancer: statistics. [Internet]. Approved by editorial board Jan 2020 [cited 28 Aug 2020]. https://www.cancer.net/cancer-types/head-and-neck-cancer/statistics.
- Bray F, Ferlay J, Soerjomataram I, Siegel RL, Torre LA, Jemal A. Global cancer statistics 2018: GLOBOCAN estimates of incidence and mortality worldwide for 36 cancers in 185 countries. CA Cancer J Clin. 2018;68(6):394–424.
- 17. Elting LS, Keefe DM, Sonis ST, Garden AS, Spijkervet FKL, Barasch A, et al. Patient-reported measurements of oral mucositis in head and neck cancer patients treated with radiotherapy with or without chemotherapy: demonstration of increased frequency, severity, resistance to palliation, and impact on quality of life. Cancer. 2008;113(10):2704–13.
- Cheng KKF, Chang AM, Yuen MP. Prevention of oral mucositis in paediatric patients treated with chemotherapy: a randomised crossover trial comparing two protocols of oral care. Eur J Cancer. 2004;40(8):1208–16.
- Cheng KK-F. Oral mucositis, dysfunction, and distress in patients undergoing cancer therapy. J Clin Nurs. 2007;16(11):2114–21.
- 20. Nonzee NJ, Dandade NA, Patel U, Markossian T, Agulnik M, Argiris A, et al. Evaluating the supportive care costs of severe

radiochemotherapy-induced mucositis and pharyngitis: results from a Northwestern University Costs of Cancer Program pilot study with head and neck and nonsmall cell lung cancer patients who received care at a county hospital, a Veterans Administration hospital, or a comprehensive cancer care center. Cancer. 2008;113(6): 1446–52.

- Carlotto A, Hogsett VL, Maiorini EM, Razulis JG, Sonis ST. The economic burden of toxicities associated with cancer treatment: review of the literature and analysis of nausea and vomiting, diarrhoea, oral mucositis and fatigue. PharmacoEconomics. 2013;31(9):753–66.
- 22. Xu X, Barron R, Mautner B, Goss T. Association between respondent- and practice-related characteristics and radiation oncology staff-reported burden on mucositis management for head and neck cancer (HNC) patients (abstract PCN96). Value Health. 2008;11(6):PA490.
- Elting LS, Cooksley CD, Chambers MS, Garden AS. Risk, outcomes, and costs of radiation-induced oral mucositis among patients with head-and-neck malignancies. Int J Radiat Oncol Biol Phys. 2007;68(4):1110–20.
- Blijlevens NMA, Donnelly JP, Yakar D, Van Die CE, De Witte T. Determining mucosal barrier injury to the oesophagus using CT scan. Support Care Cancer. 2007;15(9):1105–8.
- 25. Cartoni C, Dragoni F, Micozzi A, Pescarmona E, Mecarocci S, Chirletti P, et al. Neutropenic enterocolitis in patients with acute leukemia: prognostic significance of bowel wall thickening detected by ultrasonography. J Clin Oncol. 2001;19(3):756–61.
- World Health Organization. WHO handbook for reporting results of cancer treatment. Geneva: World Health Organization; 1979. p. 15–22.
- Cox JD, Stetz J, Pajak TF. Toxicity criteria of the Radiation Therapy Oncology Group (RTOG) and the European Organization for Research and Treatment of Cancer (EORTC). Int J Radiat Oncol Biol Phys. 1995;31(5):1341–6.
- Cancer Therapy Evaluation Program. Common Terminology Criteria for Adverse Events v3.0 (CTCAE). 9 Aug 2006. https:// ctep.cancer.gov/protocolDevelopment/electronic_applications/ docs/ctcaev3.pdf.
- Sonis ST, Eilers JP, Epstein JB, LeVeque FG, Liggett WH, Mulagha MT, et al. Validation of a new scoring system for the assessment of clinical trial research of oral mucositis induced by radiation or chemotherapy. Cancer. 1999;85(10):2103–13.
- Kushner JA, Lawrence HP, Shoval I, Kiss TL, Devins GM, Lee L, Tenenbaum HC. Development and validation of a Patient-Reported Oral Mucositis Symptom (PROMS) scale. J Can Dent Assoc. 2008;74(1):59.
- 31. Kazemian A, Kamian S, Aghili M, Hashemi FA, Haddad P. Benzydamine for prophylaxis of radiation-induced oral mucositis in head and neck cancers: a double-blind placebo-controlled randomized clinical trial. Eur J Cancer Care (Engl). 2009;18(2):174–8.
- Prada A, Chiesa F. Effects of benzydamine on the oral mucositis during antineoplastic radiotherapy and/or intra-arterial chemotherapy. Int J Tissue React. 1987;9(2):115–9.
- Hadjieva T, Cavallin-Ståhl E, Linden M, Tiberg F. Treatment of oral mucositis pain following radiation therapy for head-and-neck cancer using a bioadhesive barrier-forming lipid solution. Support Care Cancer. 2014;22(6):1557–62.
- 34. Sheibani KM, Mafi AR, Moghaddam S, Taslimi F, Amiran A, Ameri A. Efficacy of benzydamine oral rinse in prevention and management of radiation-induced oral mucositis: a double-blind placebo-controlled randomized clinical trial. Asia Pac J Clin Oncol. 2015;11(1):22–7.
- 35. Epstein JB, Stevenson-Moore P. Benzydamine hydrochloride in prevention and management of pain in oral mucositis associated with radiation therapy. Oral Surg Oral Med Oral Pathol. 1986;62(2):145–8.

- 36. Epstein JB, Silverman S, Paggiarino DA, Crockett S, Schubert MM, Senzer NN, et al. Benzydamine HCl for prophylaxis of radiation-induced oral mucositis: results from a multicenter, randomized, double-blind, placebo-controlled clinical trial. Cancer. 2001;92(4):875–85.
- Epstein JB, Stevenson-Moore P, Jackson S, Mohamed JH, Spinelli JJ. Prevention of oral mucositis in radiation therapy: a controlled study with benzydamine hydrochloride rinse. Int J Radiat Oncol Biol Phys. 1989;16(6):1571–5.
- Nomura M, Kamata M, Kojima H, Hayashi K, Sawada S. Irsogladine maleate reduces the incidence of fluorouracil-based chemotherapyinduced oral mucositis. Ann Oncol. 2013;24(4):1062–6.
- 39. Yasuda T, Chiba H, Satomi T, Matsuo A, Kaneko T, Chikazu D, Miyamatsu H. Preventive effect of rebamipide gargle on chemoradiotherpy-induced oral mucositis in patients with oral cancer: a pilot study. J Oral Maxillofac Res. 2012;2(4):e3.
- 40. Peterson DE, Ohrn K, Bowen J, Fliedner M, Lees J, Loprinzi C, et al. Mucositis Study Group of the Multinational Association of Supportive Care in Cancer/International Society of Oral Oncology (MASCC/ISOO). Systematic review of oral cryotherapy for management of oral mucositis caused by cancer therapy. Support Care Cancer. 2013;21(1):327–32.
- 41. Riley P, Glenny AM, Worthington HV, Littlewood A, Clarkson JE, McCabe MG. Interventions for preventing oral mucositis in patients with cancer receiving treatment: oral cryotherapy. Cochrane Database Syst Rev. 2015;12:CD011552.
- 42. Wilairat P, Kengkla K, Kaewpanan T, Kaewthong J, Ruankon S, Subthaweesin C, et al. Comparative efficacy and safety of interventions for preventing chemotherapy-induced oral mucositis in adult cancer patients: a systematic review and network meta-analysis. Eur J Hosp Pharm. 2020;27(2):103–10.
- 43. Spielberger R, Stiff P, Bensinger W, Gentile T, Weisdorf D, Kewalramani T, et al. Palifermin for oral Mucositis after intensive therapy for hematologic cancers. N Engl J Med. 2004;351(25): 2590–8.
- 44. Peterson DE, Boers-Doets CB, Bensadoun RJ, Herrstedt J, ESMO Guidelines Committee. Management of oral and gastrointestinal mucosal injury: ESMO clinical practice guidelines for diagnosis, treatment, and follow-up. Ann Oncol. 2015;26(Suppl 5):v139–51.
- 45. Zadik Y, Arany PR, Fregnani ER, Bossi P, Antunes HS, Bensadoun RJ, et al. Mucositis Study Group of the Multinational Association of Supportive Care in Cancer/International Society of Oral Oncology (MASCC/ISOO). Systematic review of photobiomodulation for the management of oral mucositis in cancer patients and clinical practice guidelines. Support Care Cancer. 2019;27(10):3969–83.
- 46. Clarkson JE, Worthington HV, Eden OB. Interventions for treating oral mucositis for patients with cancer receiving treatment. Cochrane Database Syst Rev. 2007 Apr 18;(2):CD001973. Update in: Cochrane Database Syst Rev. 2010;(8):CD001973.
- 47. Sarvizadeh M, Hemati S, Meidani M, Ashouri M, Roayaei M, Shahsanai A. Morphine mouthwash for the management of oral mucositis in patients with head and neck cancer. Adv Biomed Res. 2015;4:44.
- Cerchietti LCA, Navigante AH, Bonomi MR, Zaderajko MA, Menéndez PR, Pogany CE, et al. Effect of topical morphine for mucositis-associated pain following concomitant chemoradiotherapy for head and neck carcinoma. Cancer. 2002;95(10):2230–6.
- 49. Xing SZ, Zhang Y. Efficacy and safety of transdermal fentanyl for the treatment of oral mucositis pain caused by chemoradiotherapy in patients with esophageal squamous cell carcinoma. Support Care Cancer. 2015;23(3):753–9.
- 50. Bar Ad V, Weinstein G, Dutta PR, Chalian A, Both S, Quon H. Gabapentin for the treatment of pain related to radiationinduced mucositis in patients with head and neck tumors treated with intensity-modulated radiation therapy. Head Neck. 2010;32(2):173–7.

- 51. Ad VB, Weinstein G, Dutta PR, Dosoretz A, Chalian A, Both S, et al. Gabapentin for the treatment of pain syndrome related to radiationinduced mucositis in patients with head and neck cancer treated with concurrent chemoradiotherapy. Cancer. 2010;116(17):4206–13.
- 52. Leenstra JL, Miller RC, Qin R, Martenson JA, Dornfeld KJ, Bearden JD, et al. Doxepin rinse versus placebo in the treatment of acute oral mucositis pain in patients receiving head and neck radiotherapy with or without chemotherapy: a phase III, randomized, double-blind trial (NCCTG-N09C6 [Alliance]). J Clin Oncol. 2014;32(15):1571–7.
- 53. Epstein JB, Truelove EL, Oien H, Allison C, Le ND, Epstein MS. Oral topical doxepin rinse: analgesic effect in patients with oral mucosal pain due to cancer or cancer therapy. Oral Oncol. 2001;37(8):632–7.
- 54. Chattopadhyay S, Saha A, Azam M, Mukherjee A, Sur P. Role of oral glutamine in alleviation and prevention of radiation-induced oral mucositis: a prospective randomized study. South Asian J Cancer. 2014;3(1):8–12.
- 55. Tsujimoto T, Yamamoto Y, Wasa M, Takenaka Y, Nakahara S, Takagi T, et al. L-glutamine decreases the severity of mucositis induced by chemoradiotherapy in patients with locally advanced head and neck cancer: a double-blind, randomized, placebo-controlled trial. Oncol Rep. 2015;33(1):33–9.
- 56. Lin LC, Que J, Lin LK, Lin FC. Zinc supplementation to improve mucositis and dermatitis in patients after radiotherapy for headand-neck cancers: a double-blind, randomized study. Int J Radiat Oncol Biol Phys. 2006;65(3):745–50.
- 57. Moslemi D, Babaee N, Damavandi M, Pourghasem M, Moghadamnia AA. Oral zinc sulphate and prevention of radiationinduced oropharyngealmucositis in patients with head and neck cancers: a double blind, randomized controlled clinical trial. Int J Radiat Res. 2014;12(3):235–42.
- Harada K, Ferdous T, Horinaga D, Uchida K, Mano T, Mishima K, et al. Efficacy of elemental diet on prevention for chemoradiotherapy-induced oral mucositis in patients with oral squamous cell carcinoma. Support Care Cancer. 2016;24(2): 953–9.
- Khurana H, Pandey RK, Saksena AK, Kumar A. An evaluation of vitamin E and pycnogenol in children suffering from oral mucositis during cancer chemotherapy. Oral Dis. 2013;19(5):456–64.
- Wadleigh RG, Redman RS, Lou GM, Krasnow SH, Anderson A, Cohen MH. Vitamin E in the treatment of chemotherapy-induced mucositis. Am J Med. 1992;92(5):481–4.
- El-Housseiny AA, Saleh SM, El-Masry AA, Allam AA. The effectiveness of vitamin "E" in the treatment of oral mucositis in children receiving chemotherapy. J Clin Pediatr Dent. 2007;31(3):167–70.
- 62. Ferreira PR, Fleck JF, Diehl A, Barletta D, Braga-Filho A, Barletta A, et al. Protective effect of alpha-tocopherol in head and neck cancer radiation-induced mucositis: a double-blind randomized trial. Head Neck. 2004;26(4):313–21.
- 63. Tian X, Xu L, Liu X, Wang CC, Xie W, Jiménez-herrera MF, et al. Impact of honey on radiotherapy-induced oral mucositis in patients with head and neck cancer: a systematic review and meta-analysis. Ann Palliat Med. 2020;9(4):1431–41.
- 64. Im SA, Lee YR, Lee YH, Lee MK, Park YI, Lee S, et al. In vivo evidence of the immunomodulatory activity of orally administered Aloe vera gel. Arch Pharm Res. 2010;33(3):451–6.
- 65. Sahebjamee M, Mansourian A, Mohammad MHM, Zadeh MT, Bekhradi R, Kazemian A, et al. Comparative efficacy of aloe vera and benzydamine mouthwashes on radiation-induced oral mucositis: a triple-blind, randomised, controlled clinical trial. Oral Heal Prev Dent. 2015;13(4):309–15.
- 66. Puataweepong P, Dhanachai M, Dangprasert S, et al. The efficacy of oral Aloe vera juice for radiation induced mucositis in head and neck cancer patients: a double-blind placebo-controlled study. Asian Biomed. 2009;3(4):375–82.

- Nagi R, Patil DJ, Rakesh N, Jain S, Sahu S. Natural agents in the management of oral mucositis in cancer patients-systematic review. J Oral Biol Craniofac Res. 2018;8(3):245–54.
- Wilken R, Veena MS, Wang MB, Srivatsan ES. Curcumin: a review of anti-cancer properties and therapeutic activity in head and neck squamous cell carcinoma. Mol Cancer. 2011;10:12.
- 69. Saldanha SP, Almeida VD. A comparative study to assess the effectiveness of turmeric mouth wash versus saline mouth wash on treatment induced oral mucositis (TIOM) in a selected hospital at Mangalore. J Clin Res Bioeth. 2014;5:6.
- 70. Rao S, Dinkar C, Vaishnav LK, Rao P, Rai MP, Fayad R, et al. The Indian spice turmeric delays and mitigates radiation-induced oral mucositis in patients undergoing treatment for head and neck cancer: an investigational study. Integr Cancer Ther. 2014;13(3): 201–10.
- 71. Hong CHL, Gueiros LA, Fulton JS, Cheng KKF, Kandwal A, Galiti D, et al. Mucositis Study Group of the Multinational Association of Supportive Care in Cancer/International Society for Oral Oncology (MASCC/ISOO). Systematic review of basic oral care for the management of oral mucositis in cancer patients and clinical practice guidelines. Support Care Cancer. 2019;27(10):3949–67.
- 72. Komaki R, Lee JS, Kaplan B, Allen P, Kelly JF, Liao Z, et al. Randomized phase III study of chemoradiation with or without amifostine for patients with favorable performance status inoperable stage II-III non-small cell lung cancer: preliminary results. Semin Radiat Oncol. 2002;12(1 Suppl):46–9.
- Franzén L, Hellsing U, Henriksson R, Littbrand B. Managing sideeffects in radiotherapy with regard to the gastrointestinal tract. Recent Results Cancer Res. 1988;108:127–33.
- Henriksson R, Franzén L, Littbrand B. Prevention of irradiationinduced bowel discomfort by sucralfate: a double-blind, placebocontrolled study when treating localized pelvic cancer. Am J Med. 1991;91(2 Suppl 1):S151–7.
- 75. Sahakitrungruang C, Patiwongpaisarn A, Kanjanasilp P, Malakorn S, Atittharnsakul P. A randomized controlled trial comparing colonic irrigation and oral antibiotics administration versus 4% formalin application for treatment of hemorrhagic radiation proctitis. Dis Colon Rectum. 2012;55(10):1053–8.
- 76. Sahakitrungruang C, Thum-Umnuaysuk S, Patiwongpaisarn A, Atittharnsakul P, Rojanasakul A. A novel treatment for haemorrhagic radiation proctitis using colonic irrigation and oral antibiotic administration. Color Dis. 2011;13(5):e79–82.
- 77. Razzaghdoust A, Mozdarani H, Mofid B, Aghamiri SMR, Heidari AH. Reduction in radiation-induced lymphocytopenia by famotidine in patients undergoing radiotherapy for prostate cancer. Prostate. 2014;74(1):41–7.
- Topkan E, Parlak C, Topuk S, Pehlivan B. Influence of oral glutamine supplementation on survival outcomes of patients treated with concurrent chemoradiotherapy for locally advanced non-small cell lung cancer. BMC Cancer. 2012;12(1):502.
- 79. Algara M, Rodríguez N, Viñals P, Lacruz M, Foro P, Reig A, et al. Prevention of Radiochemotherapy-induced esophagitis with glutamine: results of a pilot study. Int J Radiat Oncol Biol Phys. 2007;69(2):342–9.
- 80. Jazieh AR, Younas A, Safa M, Redmond K, Buncher R, Howington J. Phase I clinical trial of concurrent paclitaxel, carboplatin, and external beam chest irradiation with glutamine in patients with

locally advanced non-small cell lung cancer. Cancer Investig. 2007;25(5):294-8.

- 81. Iyama S, Sato T, Tatsumi H, Hashimoto A, Tatekoshi A, Kamihara Y, et al. Efficacy of enteral supplementation enriched with glutamine, fiber, and oligosaccharide on mucosal injury following hematopoietic stem cell transplantation. Case Rep Oncol. 2014;7(3):692–9.
- Demers M, Dagnault A, Desjardins J. A randomized double-blind controlled trial: impact of probiotics on diarrhea in patients treated with pelvic radiation. Clin Nutr. 2014;33(5):761–7.
- Mego M, Chovanec J, Vochyanova-Andrezalova I, Konkolovsky P, Mikulova M, Reckova M, et al. Prevention of irinotecan induced diarrhea by probiotics: a randomized double blind, placebo controlled pilot study. Complement Ther Med. 2015;23(3):356–62.
- 84. Konishi T, Watanabe T, Kitayama J, Shibahara J, Nagawa H. Endoscopic and histopathologic findings after formalin application for hemorrhage caused by chronic radiation-induced proctitis. Gastrointest Endosc. 2005;61(1):161–4.
- 85. Nelamangala Ramakrishnaiah VP, Javali TD, Dharanipragada K, Reddy KS, Krishnamachari S. Formalin dab, the effective way of treating haemorrhagic radiation proctitis: a randomized trial from a tertiary care hospital in South India. Color Dis. 2012;14(7):876–82.
- 86. Bradstock KF, Link E, Collins M, Di Iulio J, Lewis ID, Schwarer A, et al. A randomized trial of prophylactic palifermin on gastrointestinal toxicity after intensive induction therapy for acute myeloid leukaemia. Br J Haematol. 2014;167(5):618–25.
- 87. Campbell P, Friebe A, Foulstone P, Grigg A, Hempton J, Bajel A. Impact of palifermin on mucosal toxicity in autologous stem cell transplants using busulfan-melphalan conditioning chemotherapy for Hodgkin and non-Hodgkin lymphoma. Leuk Lymphoma. 2012 Jul;53(7):1415–6.
- Tahir ARM, Westhuyzen J, Dass J, Collins MK, Webb R, Hewitt S, et al. Hyperbaric oxygen therapy for chronic radiation-induced tissue injuries: Australasia's largest study. Asia Pac J Clin Oncol. 2015;11(1):68–77.
- Arribas L, Hurtós L, Sendrós MJ, Peiró I, Salleras N, Fort E, et al. NUTRISCORE: a new nutritional screening tool for oncological outpatients. Nutrition. 2017;33:297–303.
- 90. Chao PC, Chuang HJ, Tsao LY, Chen PY, Hsu CF, Lin HC, Chang CY, Lin CF. The Malnutrition Universal Screening Tool (MUST) and a nutrition education program for high risk cancer patients: strategies to improve dietary intake in cancer patients. Biomedicine (Taipei). 2015;5(3):17.
- 91. Bensinger W, Schubert M, Ang KK, Brizel D, Brown E, Eilers JG, Elting L, Mittal BB, Schattner MA, Spielberger R, Treister NS, Trotti AM 3rd. NCCN Task Force Report. Prevention and management of mucositis in cancer care. J Natl Compr Cancer Netw. 2008;6(Suppl 1):S1–21; quiz S22-4.
- 92. Fromme EK, Eilers KM, Mori M, Hsieh YC, Beer TM. How accurate is clinician reporting of chemotherapy adverse effects? A comparison with patient-reported symptoms from the Quality-of-Life Questionnaire C30. J Clin Oncol. 2004;22(17):3485–90.
- Baldo P, Fornasier G, Ciolfi L, Sartor I, Francescon S. Pharmacovigilance in oncology. Int J Clin Pharm. 2018;40(4): 832–41.
- 94. Sewal RK, Saini VK, Medhi B. Forensic pharmacovigilance: newer dimension of pharmacovigilance. J Forensic Legal Med. 2015;34:113–8.



Stoma Complications

Patricia A. Brock and Kumar Alagappan

Introduction

Creation of a stoma in the course of cancer treatment is common and may lead to complications seen in the emergency department (ED). The United Ostomy Association estimates that slightly more than 500,000 Americans now have some type of stoma. Annually in the USA, roughly 100,000 people undergo the creation of either a colostomy or ileostomy. The incidence is increasing due to the rising prevalence of colorectal cancer and diverticular disease in this country [1]. More than 75% of stomas are placed as part of the management of colorectal cancer [2]. This surgically created opening placed in the abdomen is felt to be a simple procedure; however, consequences can be complex, serious, and costly to the healthcare system. Data regarding complication rates is lacking but is thought to range from 21 to 70% in one review [3], while another study reports morbidity ranging between 31 and 60% [1]. Stomas are formed in both the gastrointestinal and urogenital tracts as a means of redirecting stool and urine content, respectively. They can be permanent or temporary. Some are placed after a planned surgical procedure, while others follow an emergency operation. Surgical advancements improving ostomy care include enhanced recovery after surgery as well as new robotic and laparoscopic techniques. Additionally, ostomy equipment enhancements provide better appliance options for patients [4].

Case Study

A 49-year-old female underwent a subtotal colectomy and proctectomy with end ileostomy formation. She had initial difficulties with leakage, but follow-up with the stomal therapist led to a proper pouching system and she remained leak-free. Three weeks postoperatively, she presented to the

P. A. Brock $(\boxtimes) \cdot K$. Alagappan

Department of Emergency Medicine, MD Anderson Cancer Center, Houston, TX, USA e-mail: PABrock@mdanderson.org ED with high stomal output (4–5 L/day) and an elevated creatinine. Her abdomen was nontender. She received IV fluids in the ED and was admitted to the hospital. Stool studies were negative, and her diarrhea was controlled through dietary manipulation and addition of antidiarrheal medication.

Classification

Intestinal ostomies are classified according to the segment of the bowel brought to the surface. Ileostomies or small-bowel ostomies are often located on the left side of the abdomen. In the USA, an estimated 165,000–265,00 individuals live with this type of ostomy at the rate of 40,000 new ostomies annually [5]. This is distinguished from colostomies, which originate from the large bowel and are most often on the left side of the abdomen. Ostomies can either be a terminal or end ostomy with the bowel divided and the proximal stump brought to the surface. Alternatively, a loop ostomy is created when the anterior wall of the bowel is brought to the surface and is not transected.

A urostomy is made in cases where drainage of urine through the bladder and urethra is not possible, as in the case of a cystectomy following the diagnosis of invasive bladder cancer, the sixth most common cause of cancer in the USA.

Factors such as body habitus and pre-existing conditions contribute to poor outcomes. Although most issues arise in the first 5 years postoperatively, the risk of complications from stoma formation can last throughout one's life and can affect daily activities for the patient [4]. Early complications include inappropriate placement, skin irritation, soilage, stoma retraction, dehydration, and stoma necrosis. Late complications include stomal prolapse, peristomal hernia, chronic dermatitis, and orifice narrowing. Compounding ostomy complications in cancer patients are issues related to ongoing treatments with chemotherapy, immunotherapy, and radiation, as well as the increased prevalence of malnutrition and weight loss.

© Springer Nature Switzerland AG 2021

K. H. Todd et al. (eds.), Oncologic Emergency Medicine, https://doi.org/10.1007/978-3-030-67123-5_51

When evaluating the ostomy/stoma, physicians should be familiar with common complications and be ready to call for a surgical consult, if needed, especially if unfamiliar with the device.

Health Economics

A 2017 study by Taneja et al. estimated the cost of peristomal skin complications in the USA. In their retrospective cohort of 128 patients undergoing colostomy, ileostomy or urostomy (cancer and non-cancer related), they found that one-third experienced peristomal skin complications within 90 days and that total healthcare costs were almost \$80,000 higher for these patients in comparison with those without such complications [6]. In addition to economic costs, cancer survivors with ostomies face a number of challenges related to self-care. Bulkley et al. surveyed 177 long-term rectal cancer survivors with ostomies, finding that the majority reported ongoing challenges, most commonly leakage around the pouching system and peristomal skin problems [7]. In a more recent study, Sun et al. found that the greatest challenges posed by ostomies were associated with the use of ostomy appliances and time requirements for ostomy care [8]. Additionally, they noted bleeding, pain, leakage, skin problems, and ostomy cleaning as challenges. Patients reported anxiety related to leakage, odor, and skin irritation as barriers to social activity and self-confidence. To prevent ostomy-related ED visits, enhancements in communitybased, ostomy self-management initiatives may hold promise. In 2018, Hornbrook et al. estimated the cost of an ostomy nurse-led, small-group program, and concluded that a fivesession training course conducted over 8 weeks would cost \$1812 per patient [9]. Given the high cost of stoma complications, this expense would appear to provide a cost benefit.

Diarrhea

Daily volumes exceeding one liter above baseline generally define the presence of ileostomy-associated diarrhea. When portions of the ileum and the colon are not available for normal fluid absorption, excess liquid and the accompanying electrolytes are eliminated through the ileostomy. Adaptive changes occur after the initial postoperative period in many patients, but in some, this condition, also referred to as high-output stoma, is troublesome and can be dangerous [5]. High-output stomas occur in 16% of stomas created [10]. In the early postoperative period, 16–50% of patients report high-output diarrhea, and nearly 20% require readmission as a consequence [5].

When presenting to the ED, this complaint is often associated with dehydration and electrolyte imbalances. Close attention to renal function is necessary as having an ileostomy is a risk factor for kidney failure [5]. Large sodium losses with high volumes of fecal waste may occur, leading to hyponatremia. Poor absorption of magnesium also occurs. Evaluation for an obstruction or an infectious etiology, most commonly, *Clostridium difficile*, is important. Risk factors for high stomal output include the use of diuretics, coexisting diabetes mellitus, and having undergone a total proctectomy [10].

Diarrhea associated with a colostomy is described as having loose or watery bowel movements more than four times in 1 day. Although benign causes such as a change in diet may be the culprit, more severe etiologies include infection or an intestinal blockage.

After a thorough medical history, noting prior surgeries and oncologic interventions, the emergency evaluation of a patient presenting with diarrhea will include laboratory analysis of the electrolytes and renal function, as well as close attention to the volume status of the patient, which may be manifested by a low blood pressure or an elevated heart rate. Skin turgor and mucus membranes should be noted, and palpation and auscultation of the abdomen may detect tenderness or signs of obstruction. The clinician should directly examine the stoma and its contents. Stool analysis for an infectious process, most commonly, C. difficile, is important. A review of all medications for prokinetic properties, including metoclopramide, laxatives, erythromycin, etc., is imperative. Metformin may provoke increased stomal output, as can an abrupt discontinuation of steroids. Review of dietary changes may be informative. Imaging should be ordered only if an obstructive etiology is considered or pain is a prominent feature of the presentation.

Initial management should focus on fluid management, electrolyte replacement, and control of the diarrhea. If tolerated, fluid resuscitation may be provided via the oral route, with isotonic drinks the best option. Avoid intake of hypotonic drinks, tea, coffee, alcohol, and fruit juices. In most cases, however, adequate venous access will be important. IV supplementation of electrolytes will be guided by laboratory results. Cardiac monitoring may be necessary in the setting of severe electrolyte derangements. Most commonly used antimotility agents include diphenoxylate/atropine and loperamide. In several randomized studies of established ostomies, loperamide reduced output by 22–30% [5]. The typical dose is 4 mg four times per day before meals and at bedtime.

In severe cases where outputs are over 3–4 L per day, octreotide can be used [5]. However, if considering octreotide, the patient will likely require a general surgery consultation and admission.

If available, consultation with an experienced enterostomal therapist could provide assurance of a good device fit and reinforce the patient's knowledge of daily maintenance. Decisions for inpatient admission or observation are made based on labs, volume status, and whether the patient tolerates fluids orally.

Skin Irritation

Despite the improvement of ostomy systems and study of the creation and care of the stoma, skin complications remain a common occurrence following colostomy, ileostomy, or urostomy surgery. Its occurrence is reported to be 43%, most commonly after an ileostomy [11]. Skin irritation is related to the caustic, watery content of the frequent movements, which contain proteolytic enzymes and a high pH [12]. Illfitting appliances leading to areas of the skin exposed to erosive content may be a consequence of normal ostomy maturation, and the remedy may be as easy as proper adjustment of wafer size [13]. However, prevention of this complication should begin in the operating room with creation of a quality stoma protruding 2-3 cm above the skin surface [4]. Additionally, poor placement of the ostomy in relation to body folds makes secure application of the appliance difficult. Regular assessment of the pouching system may avert the skin irritations that can become cyclic if left unchecked. Infectious causes, mainly fungal, also result in skin irritation and are seen in immunocompromised and diabetic patients (Fig. 51.1) [14].

Visual inspection of the intact ostomy bag is key, noting whether the selected wafer size matches the ostomy. Examination of the skin may reveal an erythematous papular rash consistent with candidiasis. This rash may be present under the wafer as well as in the skin surrounding the stoma. An allergic contact dermatitis, often due to an ostomy care product, may produce papules, vesicles, and redness. The patient will complain of itching and burning.

Ensuring proper wafer placement is often aided by a referral to an ostomy specialist. To manage suspected contact dermatitis, change the product currently being used, add a skin sealant to the local area, and make an outpatient referral to a stomal specialist or a dermatologist. Irritant dermatitis responds well to topical corticosteroid lotions [13]. Nystatin powder is the optimal treatment when candidiasis is suspected.

Hernia

A parastomal hernia (PSH) can occur with either an ileostomy or colostomy [15] and is variably reported to have a



Fig. 51.1 Peristomal dermatitis. Irritation caused by the effluent in an inadequate pouch adaptation to the skin allowing the prolonged feces/ skin contact (\mathbf{a}) and an early pouch detachment (\mathbf{b}). Blister at the adhesive area in the periphery of pouch resin itself (\mathbf{c}). Dermatitis caused by

both pouch resin and peripheral adhesive (**d**). Dermatitis due to contact of feces with the skin (**e**). Fungal dermatitis (**f**). (From Rodrigues et al. [14] with permission CC by 3.0)



Fig. 51.2 Parastomal hernia in a prolapsed stoma (a) and a huge parastomal hernia (b). (From Rodrigues et al. [14] with permission CC by 3.0)

3–50% incidence due to a wide range of definitions [10]. Most are asymptomatic, but as they grow these hernias can cause problems. End colostomy has a higher rate of parastomal hernias. A late complication of ostomy surgery, PSH is symptomatic in 75% of patients but can lead to a reduced quality of life. Symptoms range from esthetic complaints, discomfort, local pain, leakage, and poor appliance application to emergent presentations with strangulation and obstruction. Risk factors for the development of a hernia are either patient-related or attributed to a technical problem. Perioperative steroid use has also been cited as a factor [16] (Fig. 51.2).

Clinical examination is necessary for the diagnosis of parastomal hernia. Direct inspection can be challenging, as detection of a bulge and cough impulse is a subjective finding. Details of the fascial defect and concomitant incisional hernias, as well as other occult pathologies, may obscure the diagnosis. In addition, clinical detection of PSH can be challenging in the obese patient. When diagnostic uncertainty exists, computed tomography and ultrasonography may be useful adjuncts. Nonurgent surgical evaluation is advised, unless a surgical emergency is suspected.

The majority of peristomal hernias are managed conservatively with the aid of a stoma care nurse advising the patient on proper appliance placement and improved adhesive adjuncts, as well as supportive garments. Outpatient referral for elective surgical intervention should be considered for those with high symptom burden. However, the patient presenting with a picture of bowel obstruction and/or signs of intestinal ischemia requires expeditious surgical consultation and resuscitation while preparing for a possible eminent operative procedure.

Retraction

Although usually seen acutely in the immediate postoperative period due to ischemic changes, retraction can be a late stoma complication. Retraction is the inversion of the mucocutaneous junction toward the abdominal wall. Poor fitting of the stoma appliance results in spillage of stool contents and skin irritation/inflammation. The latter is likely the reason for ED presentation (Fig. 51.3). In mildly symptomatic patients, a change in the stomal appliance may be useful to decrease bowel leakage. In severely symptomatic patients, a surgical referral for revision is necessary.

Stenosis

A stenosis or stricture of an ostomy can cause a mechanical obstruction to the passage of bowel contents. The patient presents with symptoms ranging from constipation to crampy abdominal pain. Ischemia of the bowel at the level of stoma may be causative; in patients with a prior history of Crohn's or malignancy, this may represent a recurrence. A narrowing of a colostomy may result from local infection or inadequate skin opening. Strictures associated with ileostomies most often require definitive surgical treatment. Those associated


Fig. 51.3 Ostomy retraction with light (a) and severe (b) inflammatory reaction. (From Rodrigues et al. [14] with permission CC by 3.0)



Fig. 51.4 Stoma stenosis. Note the retraction of the skin (**a**) and the scar tissue around the stoma opening in a chronic stoma stenosis (**b**). (From Rodrigues et al. [14] with permission CC by 3.0)

with colostomies may respond to dietary modifications and expectant management (Fig. 51.4).

In addition to a thorough abdominal assessment, examination of the stoma is critical. Digital inspection to the level of the fascia will yield important information regarding the degree of stenosis. Imaging will be required if obstruction is suspected.

Minor stenosis can be managed by changes in diet. However, symptomatic patients with obstructive symptoms require a surgical consultation and management. Management may include dilatation of the stoma with a 36 Fr Foley as a temporizing maneuver.

Prolapse

Stomal prolapse occurs as a late complication with the patient presenting to the ED with a segment of bowel protruding through the stomal opening. Occurring more often in the transverse loop colostomies [10], a prolapse can occur with any stoma and has a reported incidence between 2–26% [16]. This complication is found more often in obese patients and may be due to the original surgical technique. Although this is an unsightly and distressing complication, it is usually of little clinical significance. However, left untreated it can result in incarceration and strangulation from edematous bowel (Fig. 51.5).

Symptoms include skin irritation, difficulty with appliance application, and bleeding from the mucosa secondary to exposure. There is a low risk of bowel obstruction, incarceration, and strangulation associated with stomal prolapse.

The patient's history of an acute onset versus intermittent prolapse will supplement the necessary digital inspection of the ostomy for bowel viability. Further workup for bowel strangulation or obstruction should be conducted as indicated.

Some prolapses are amenable to bedside reduction. With significant bowel edema, however, osmotic therapy with



Fig. 51.5 Stoma prolapse. Note the abnormal length of the stoma. If left untreated the ostomate is more susceptible to abrasions or infection. (From Rodrigues et al. [14] with permission CC by 3.0)

simple table sugar applied to the prolapsed stoma and allowed to sit for 30 minutes may allow for a reduction [10]. Outpatient referral can be considered for further evaluation along with adjustment in pouch size. If viability of the bowel is in question, a surgical evaluation is necessary in the ED.

Bleeding

Local stomal bleeding is a rare occurrence. Local trauma can cause visible blood related to improper pouching technique or a rigid appliance encroaching on the mucosa. Recurrent trauma can lead to exuberant granulation tissue that is friable and bleeds frequently. A more consequential reason for profuse bleeding may be stomal or parastomal varices. Although usually appearing in the presence of portal hypertension, blockage of vessels within the mesentery can result in mucosal venous congestion and bleeding. In patients with a diagnosis of cirrhosis, this is a recognized ectopic site for variceal development. In some cases, these varices can be singular causes for focal bleeding. Variceal bleeding is usually painless, profound, and recurrent [17].

While digital examination of the stoma is of prime importance for local bleeding, consideration of a source anywhere along the gastrointestinal tract should be entertained and dealt with in a similar manner as in any patient with a GI bleed. Most bleeding from a stoma will have a visible site of injury or a bleeding vessel. If this is not evident, treat the patient as if they were having a GI bleed. The majority of cases of peristomal varices can be easily identified on physical exam by a bluish hue in the peristomal skin and presence of caput medusae in the peristomal area [16].

Local manual pressure is the first therapeutic maneuver with acute stomal bleeding. A useful adjunct is an epinephrine-soaked gauze. Isolated minor trauma-related bleeding will stop with local pressure and time. Granulation tissue associated with chronic local trauma is treated with local destruction by silver nitrate or judicious electrocautery. Suturing of a visible vessel can be considered, but ligation of a bleeding varix is often futile and long-term success unlikely. Mortality is high due to underlying severe liver disease. Medical and/or surgical consultation should be obtained which will focus on reduction of portal pressure.

Urostomy Complications

Urinary diversion following cystectomy for bladder cancer management is increasing in prevalence [18]. When a urostomy is created, it is either incontinent or continent, and the patient will be familiar with the type constructed. The incontinent urostomy drains into a stoma bag attached via an appliance. The continent urostomy requires intermittent catheterization in order to drain the urine.

The ileal conduit, an incontinent urostomy, is the most common type of urinary diversion following cystectomy, accounting for 85% [19]. Typically created from the terminal ileum, these stomas have a lower risk of electrolyte imbalance.

The continent urostomy uses either the ileum or colon, or a combination of both, as a reservoir. Patients are instructed to routinely self-catheterize to avoid complications such as urinary tract infections, hydronephrosis leading to upper pole injury, stone formation, and perforation.

Metabolic derangements commonly occur in patients with intestinal urinary diversions due to the absorptive surface of the bowel.

Future Needs

The creation of a stoma represents a major life event that will affect a rising number of individuals, especially as the rate of colorectal cancer accelerates. The healthcare system will be impacted, and the ostomate will continue to face the potential of complications throughout their life.

Several innovative strategies to mitigate these effects might include:

1. *Prehabilitation* is a preoperative tactic to identify and address the malnutrition and physical deconditioning present in cancer patients as well as identify psychological needs of those requiring an ostomy [20]. Adding a

robust and comprehensive discharge plan with patient education material [21] combined with referral to a support group and a connection to a stomal expert would be part of this process.

- Centralizing information regarding optimal practice patterns, including knowledge about basic ostomy care, ostomy product availability, and standardized care plans, has been advocated to improve clinical outcomes [22].
- 3. Proactive monitoring of *patient-reported outcome measures* (PROMs) has been utilized, leading to improved adaptation to life after ostomy surgery. PROMs may play a role in physical and psychological adaptation [23].
- 4. Adding *ostomy telehealth* services for cancer survivors could expand the options for meaningful follow-up for ostomates, especially new ones, whose use of resources outstrips other colorectal patients [24, 25].

References

- Sheetz KH, Waits SA, Krell RW, Morris AM, Englesbe MJ, Mullard A, et al. Complication rates of ostomy surgery are high and vary significantly between hospitals. Dis Colon Rectum. 2014;57(5):632–7.
- Ambe PC, Kurz NR, Nitschke C, Odeh SF, Möslein G, Zirngibl H. Intestinal ostomy. Dtsch Arztebl Int. 2018;115(11):182–7.
- Shabbir J, Britton DC. Stoma complications: a literature overview. Color Dis. 2010;12(10):958–64.
- Carlsson E, Fingren J, Hallén AM, Petersén C, Lindholm E. The prevalence of ostomy-related complications 1 year after ostomy surgery: a prospective, descriptive, clinical study. Ostomy Wound Manage. 2016;62(10):34–48.
- Rowe KM, Schiller LR. Ileostomy diarrhea: pathophysiology and management. Proc (Bayl Univ Med Cent). 2020;33(2):218–26.
- Taneja C, Netsch D, Rolstad BS, Inglese G, Lamerato L, Oster G. Clinical and economic burden of peristomal skin complications in patients with recent ostomies. J Wound Ostomy Continence Nurs. 2017;44(4):350–7.
- Bulkley JE, McMullen CK, Grant M, Wendel C, Hornbrook MC, Krouse RS. Ongoing ostomy self-care challenges of long-term rectal cancer survivors. Support Care Cancer. 2018;26(11):3933–9.
- Sun V, Bojorquez O, Grant M, Wendel CS, Weinstein R, Krouse RS. Cancer survivors' challenges with ostomy appliances and self-management: a qualitative analysis. Support Care Cancer. 2020;28(4):1551–4.
- Hornbrook MC, Cobb MD, Tallman NJ, Colwell J, McCorkle R, Ercolano E, et al. Costs of an ostomy self-management training program for cancer survivors. Psychooncology. 2018;27(3):879–85.

- Tsujinaka S, Tan KY, Miyakura Y, Fukano R, Oshima M, Konishi F, Rikiyama T. Current management of intestinal stomas and their complications. J Anus Rectum Colon. 2020;4(1):25–33.
- Murken DR, Bleier JIS. Ostomy-related complications. Clin Colon Rectal Surg. 2019;32(3):176–82.
- Colwell JC, Kupsick PT, McNichol LL. Outcome criteria for discharging the patient with a new ostomy from home health care: a WOCN Society Consensus Conference. J Wound Ostomy Continence Nurs. 2016;43(3):269–73.
- Lyon CC, Smith AJ, Griffiths CE, Beck MH. Peristomal dermatoses: a novel indication for topical steroid lotions. J Am Acad Dermatol. 2000;43(4):679–82.
- Rodrigues FP, Novaes JAV, Pinheiro MM, Martins P, Cunha-Melo JR. Intestinal ostomy complications and care. In: Mero V, editor. Gastrointestinal stomas. London UK: IntechOpen; 2019. https:// doi.org/10.5772/intechopen.85633.
- MacQueen IT, Kirchhoff P, Chen DC. Prevention of Parastomal Hernia. Surg Technol Int. 2018;33:sti33/1055. Epub ahead of print.
- Husain SG, Cataldo TE. Late stomal complications. Clin Colon Rectal Surg. 2008;21(1):31–40.
- Saad WE, Saad NE, Koizumi J. Stomal varices: management with decompression tips and transvenous obliteration or sclerosis. Tech Vasc Interv Radiol. 2013;16(2):176–84.
- Morrison CD, Kielb SJ. Use of bowel in reconstructive urology: what a colorectal surgeon should know. Clin Colon Rectal Surg. 2017;30(3):207–14.
- Gore JL, Litwin MS, Urologic Diseases in America Project. Quality of care in bladder cancer: trends in urinary diversion following radical cystectomy. World J Urol. 2009;27(1):45–50.
- Silver JK, Baima J, Mayer RS. Impairment-driven cancer rehabilitation: an essential component of quality care and survivorship. CA Cancer J Clin. 2013;63(5):295–317.
- Prinz A, Colwell JC, Cross HH, Mantel J, Perkins J, Walker CA. Discharge planning for a patient with a new ostomy: best practice for clinicians. J Wound Ostomy Continence Nurs. 2015;42(1):79–82.
- 22. Turnbull GB, Arnold A, Aronson L, Hawke G, LeBlanc K, Parslow N, et al. The role of industry in improving quality of life for persons with an ostomy: a Canadian consensus. Ostomy Wound Manage. 2004;50(9):78–85.
- 23. Indrebø KL, Aasprang A, Olsen TE, Andersen JR. A new model of patient-reported outcome monitoring with a clinical feedback system in ostomy care: rationale, description and evaluation protocol. Health Qual Life Outcomes. 2020;18(1):12.
- 24. Sun V, Ercolano E, McCorkle R, Grant M, Wendel CS, Tallman NJ, et al. Ostomy telehealth for cancer survivors: Design of the Ostomy Self-management Training (OSMT) randomized trial. Contemp Clin Trials. 2018;64:167–72.
- White T, Watts P, Morris M, Moss J. Virtual postoperative visits for new ostomates. Comput Inform Nurs. 2019;37(2):73–9.

Hematopoietic Cell Transplant

Sherilyn A. Tuazon, Victor A. Chow, Phuong T. Vo, and Masumi Ueda

Introduction

Hematopoietic cell transplantation (HCT) has the potential to cure both malignant and nonmalignant hematologic diseases that may not be possible with conventional therapy. HCT allows for the administration of high-dose chemotherapy and radiotherapy while providing an opportunity for hematologic recovery with stem cell reinfusion. In the context of allogeneic HCT, an immunologic attack against the tumor cells is also conferred, termed graft-versus-tumor (GVT) effect.

Broadly, HCT consists of three phases: conditioning, stem cell infusion; and, for allogeneic HCT, graft-versushost disease (GVHD) prophylaxis. The choice of conditioning regimen, stem cell source, and GVHD prophylaxis depends on patient and disease characteristics, as well as donor availability. The purpose of conditioning regimen is to eliminate tumor cells resistant to conventional doses of chemotherapy and, in the case of allogeneic HCT, produce sufficient immunosuppression to prevent graft rejection. Conditioning regimens used prior to autologous HCT are myeloablative, characterized by irreversible marrow aplasia requiring stem cell rescue. For allogeneic HCT, conditioning regimens can be classified as myeloablative; nonmyeloablative, which cause minimal cytopenias and can be given without stem cell support; and reduced-intensity, which result in intermediate degrees and duration of marrow suppression [1]. The source of stem cells can be from bone marrow, peripheral blood, or umbilical cord blood. HCT can also be categorized according to the relationship between the donor and recipient [autologous, syngeneic, or allogeneic (matched related, matched unrelated, or haploidentical)].

In 2018, approximately 23,000 HCTs were performed in the Unites States; of these, over 60% were autologous and

the remainder were allogeneic. The most common indication for autologous HCT is multiple myeloma followed by lymphoma (non-Hodgkin and Hodgkin). Acute leukemias (acute myeloid leukemia, acute lymphocytic leukemia) and myelodysplastic syndrome (MDS) are the most common indications for allogeneic transplants, accounting for 75% of allogeneic HCTs [2]. Nonmalignant conditions that can be treated with HCT include, but not limited to, aplastic anemia, thalassemia, and autoimmune disorders (e.g., systemic sclerosis).

Complications of High-Dose G-CSF Administration

The most commonly used agent for hematopoietic cell mobilization and collection is granulocyte colony-stimulating factor (G-CSF), which triggers neutrophil elastase and cathepsin G to cleave bone marrow adhesion molecules, such as vascular cell adhesion molecule-1, stromal cell-derived factor, and CXCR4 [3, 4]. It is administered generally for 4 days before apheresis, with the goal of mobilization and increasing the total white blood cell (WBC) in the peripheral blood, of which a fraction is CD34+ stem cells. The majority of adverse events (AEs) associated with G-CSF are minor and not life-threatening, including bone pain, particularly along the back, pelvis, and/or ribs (82-94%); headache (34-70%); body aches; fatigue; and nausea/vomiting; however, serious and life-threatening AEs have been reported, including splenic rupture and ischemic complications, which constitute oncologic emergencies [4].

Splenic Rupture

The Research on Adverse Drug Events and Reports (RADAR) project reviewed AEs associated with G-CSF administration [5]. In addition to the common toxicities mentioned above, they also reported rare instances of



S. A. Tuazon $(\boxtimes) \cdot V$. A. Chow $\cdot P$. T. Vo $\cdot M$. Ueda

Division of Medical Oncology, University of Washington School of Medicine, Clinical Research Division, Fred Hutchinson Cancer Research Center, Seattle, WA, USA e-mail: stuazon@fredhutch.org

splenic rupture, allergic reactions, flares of underlying autoimmune disorders, lung injury, and vascular events [5]. Several studies have evaluated short-term administration of G-CSF on spleen size, with an average increase of 10% in volume and 11 mm in length occurring in 95% of subjects [6]. Although spleen size returned to baseline ~10 days after completion of G-CSF, instances of splenic rupture have been reported in the literature. While rare, a small number of case reports note a timing of rupture between day 3 and day 10 after the first G-CSF administration with outcomes ranging from those requiring conservative measures or splenectomy and even death [5]. An open-label study evaluating splenic changes in normal donors found that a predicted twofold change in spleen volume occurred in donors 60 years of age or older, although the physiological changes accounting for this phenomenon are unknown [6]. The leading hypothesis comes from pathologic evaluation of ruptured spleens requiring splenectomy, with histology indicating extramedullary myelopoiesis possibly leading to expansion of the hematopoietic tissue in the spleen.

Ischemic Complications

Ischemic complications have also been reported after G-CSF administration, especially in individuals with known coronary artery disease (CAD). In a study by the American College of Cardiology aiming to mobilize progenitor cells from the bone marrow to promote myocardial neovascularization and relief of ischemia, no objective evidence of cardiac benefit was found; rather, there were concerns for potential adverse outcomes. Their study of 16 patients reported 2 patients with serious AEs:

- A 52-year-old diabetic man with a history of 2 myocardial infarctions (MIs) and class III angina, despite prior percutaneous coronary intervention (PCI), who developed severe chest pain, nausea, and diaphoresis with electrocardiogram (ECG) showing new ST-segment depression and mild ST-segment elevation treated with enoxaparin and nitroglycerin paste.
- 2. A 69-year-old man with class IV angina, history of MIs and 16 PCIs, coronary artery bypass, transmyocardial laser revascularization, and enhanced external counterpulsation, who experienced 3 episodes of angina without ECG changes, was ultimately hospitalized 2 weeks later for severe chest pain, and developed electrical-mechanical dissociation followed by asystole and death [7]

These individuals are extreme examples of baseline cardiovascular disease and would unlikely meet criteria for an ASCT; however, even rigorous testing with echocardiograms and ejection fraction cutoffs with input from cardiology consultation may not always identify patients at high risk for cardiovascular complications during the transplant process. Hence, it is standard practice at the Fred Hutchinson Cancer Research Center (FHCRC) to admit patients for G-CSF mobilization on telemetry if they have known CAD and a history of cardiac disease and, particularly, AL amyloidosis patients who may or may not have cardiac involvement. The pathophysiology behind exacerbations of underlying CAD by G-CSF is unclear, but in vitro experiments have shown that pro-inflammatory markers such as C-reactive protein (CRP) may play a role in promoting plaque destabilization and rupture.

Complications of Apheresis

Once the level of CD34+ cells is adequate in the peripheral blood, the process of apheresis is initiated for collection and storage of these progenitor cells. In general, apheresis is relatively safe but can also lead to serious, albeit manageable, complications. One of the most commonly encountered complications is citrate-induced hypocalcemia, as it is used as an anticoagulant throughout the apheresis procedure. Citrate will complex calcium, resulting in hypocalcemia in upward of 9% of treatments and leading to symptoms of perioral or distal extremity paresthesias [8]. If not reversed, severe citrate-induced hypocalcemia may prolong the QTc and lead to cardiac arrhythmias, in addition to muscle spasm, chest pain, and hypotension. Infusion providers are aware of this common complication and readily supply calcium tablets or intravenous calcium to mitigate hypocalcemia. Other commonly encountered phenomena after apheresis are thrombocytopenia and anemia, which may be further pronounced in a patient who recently received cytotoxic therapy or with bone marrow infiltration from their underlying disease. Various mechanisms for thrombocytopenia may contribute, and decreases of platelet count up to 50% have been reported, especially with older apheresis devices. Coupled with anemia and a hematocrit drop of up to 10%, hemorrhage may occur after an apheresis procedure [8]. Providers should be aware of these transient decreases in blood counts, monitor levels accordingly, and have a low threshold to work up any evidence for vital sign changes and signs or symptoms reminiscent of hemorrhage. Patients should be counseled on the potential need for transfusions even without bleeding events.

DMSO Toxicity

Dimethyl sulfoxide (DMSO) is the most commonly used cryopreservation agent for autologous stem cells collected via apheresis, protecting cells from the injurious effects of freezing and crystal formation prior to storage. While relatively nontoxic to patients in small amounts, reinfusion of DMSO with the thawed autologous stem cells may trigger complications as much as 20-50% of the time in the acute setting, ranging from uncomplicated to serious toxicities [9]. Manageable or expectant side effects can be seen such as skin flushing, nausea, vomiting, diarrhea, and abdominal cramping brought about by a vagal response upon intravenous infusion of the cold product; however, sudden and severe toxicities have also been reported [9, 10]. These include cardiovascular or respiratory complications such as sudden and severe hypotension, bradycardia, fatal arrhythmias, dyspnea, respiratory arrest, and diffuse alveolar hemorrhage [9, 10]. Serious neurologic complications such as reversible leukoencephalopathy, strokes, and epileptic seizures have also been reported [11]. These serious toxicities are presumably from histamine-induced vasodilation and release; therefore, DMSO toxicity may result in increased morbidity and prolonged hospital admissions and incur additional treatment-related costs. In a survey from 444 European Bone Marrow Transplantation (EBMT) centers to assess DMSO-related complications, 95 centers completed the questionnaire for a 22% response rate. Of the 95 responding centers, 57 reported at least 1 DMSO toxicity (60%). Approximately 34,000 transplants were carried out by the 95 centers, giving an overall incidence of 1.4%, or approximately 1 in every 70 transplants [12].

Case Study

A 64-year-old male with IgG kappa multiple myeloma ultimately receives his cryopreserved autologous peripheral blood stem cells and experienced no major complications; however, he now presents with vomiting, debilitating oropharyngeal pain, and diarrhea 6 days after stem cell infusion with a conditioning regimen consisting of melphalan 200 mg/ m². The patient was treated supportively with antiemetics, loperamide (after a negative stool Clostridium difficile test), intravenous fluids, patient-controlled analgesia, and total parenteral nutrition.

By day + 14, the patient noted improvement in symptoms with supportive management. However, on day + 16, the patient developed new-onset generalized erythematous rash, fever, and 3-kg weight gain. CBC with differential count showed a hemoglobin of 8.6 g/dL, white blood cell count of 1.5 K/uL, absolute neutrophil count (ANC) of 540/uL, and platelets of 19 K/uL. Comprehensive metabolic panel is remarkable only for hypokalemia at 3.6 mEq/L and slightly elevated creatinine at 1.3 mg/dL (baseline 0.8 mg/dL). Blood and urine cultures were negative. Chest X-ray showed slightly increased interstitial markings but no consolidation.

Oral Complications of HCT

Oropharyngeal mucositis (OM) is a common complication that typically peaks 6–11 days following high-dose conditioning regimens and then eventually heals over 1–2 weeks [13]. Recombinant keratinocyte growth factor (palifermin) is occasionally given with conditioning regimens, particularly those that include total body irradiation (TBI), to reduce the incidence of severe OM [14, 15]. Adverse reactions with palifermin include rash, oral dysesthesia, tongue thickening and discoloration, as well as taste alteration. With high-dose melphalan conditioning, oral cryotherapy (use of ice chips) is instituted to reduce the severity and duration of OM [16].

The development of OM may predispose patients to systemic bacterial infections originating from the mouth flora (e.g., Streptococcus viridans or S. mutans). In addition, when severe, OM may increase the risk for aspiration and airway compromise [13]. Bleeding may occur with OM, but is often minimal given platelet-support protocols and routine herpes simplex virus (HSV) prophylaxis [13]. Treatment is supportive with regular oral hygiene, saline rinses, mucosal coating agents (e.g., antacids), and topical anesthetics. Caution is advised when recommending topical anesthetics due to the risk of inadvertent mucosal injury while eating or performing oral care while the mouth is temporarily anesthetized. Oral anesthetics may also impair the gag reflex, increasing the risk of aspiration. When local measures are inadequate in providing relief, systemic pain management with opioid patient-controlled analgesia is usually effective.

Concurrent infection can accentuate the severity of OM. Candidiasis and HSV are common oral infections after HCT. However, the routine use of antifungal and antiviral prophylaxis has dramatically reduced the incidence and extent of these infections. When suspected, swabbing of the oral ulcers or lesions for histopathologic examination and culture should be performed.

Parotiditis is another oral complication typically seen among patients who are receiving total body irradiation and high-dose chemotherapy [17, 18]. Treatment is supportive with analgesics, warm compresses, and stimulation of salivation (i.e., lemon drops).

Gastrointestinal Complications of HCT

Following most conditioning regimens, nausea, vomiting, and diarrhea are common. Adequate control of nausea and vomiting is important not only for patient comfort but also to prevent complications such as dehydration, malnutrition, electrolyte or metabolic derangements, aspiration, erosive esophagitis, and esophageal tears.

Although chemotherapy-induced diarrhea is common after conditioning, patients should be ruled out for C. difficile, which is the most common cause of diarrhea in healthcare settings [19]. Because C. difficile colonization occurs in approximately 10-15% of HCT patients [20] and current testing is not capable of differentiating true disease from colonization, it is important to carefully select patients appropriate for testing. C. difficile testing is indicated in patients with new-onset diarrhea defined as three or more loose stools in 24 hours [19]. The preferred drug for treating an initial episode of C. difficile-associated diarrhea is vancomycin (125 mg PO QID \times 10 days) because it has been shown to result in superior clinical cure compared to metronidazole [19]. However, metronidazole (500 mg PO TID × 10 days) is an acceptable alternative if oral vancomycin is not available. In patients with significant diarrhea despite active C. difficile treatment, loperamide may be considered as an adjunct, but should be approached with caution because there have been no prospective or randomized studies supporting its safety in the context of active C. difficile infection [19, 21, 22].

During the expected period of profound neutropenia following HCT, patients can develop neutropenic enterocolitis (NEC) (also termed typhlitis, ileocecal syndrome, or necrotizing enterocolitis). It is manifested as fever, nausea, vomiting, diarrhea, and abdominal pain [23]. The pathogenesis likely involves chemotherapy-induced disruption of the gastrointestinal mucosa and transmural translocation of pathogens [24]. Polymicrobial infection is frequent with gram-negative bacilli, gram-positive cocci, anaerobes (C. septicum), and even fungal infection, most commonly by *Candida* spp. The diagnosis is based on clinical presentation and radiologic findings. Contrast CT can demonstrate bowel thickening or dilation, mesenteric stranding, and pneumatosis [25]. Management includes supportive therapy with bowel rest, intravenous fluids or parenteral nutrition (if indicated), and antimicrobial therapy. Empiric antibiotics should cover enteric gram-negative and anaerobic pathogens. Piperacillin-tazobactam (Zosyn) or dual therapy with cefepime plus metronidazole is preferred. The carbapenems (imipenem-cilastatin and meropenem) are alternative agents for patients allergic to the recommended antibiotics or suspected to have resistant gram-negative bacteria, such as an extended-spectrum beta-lactamase (ESBL)-producing organisms [26].

Engraftment Syndrome

Neutrophil engraftment is defined as an ANC \geq 500/uL for three consecutive days, whereas platelet engraftment is defined as an unsupported platelet count \geq 20,000/uL. The timing for neutrophil engraftment typically occurs 2–3 weeks

after stem cell reinfusion, but largely varies on patient and disease characteristics, stem cell source, and dose and conditioning regimen.

Engraftment syndrome (ES) can occur as a complication of autologous HCT, approximately 4–5 days before and after neutrophil recovery [27, 28]. ES is likely mediated by the release of pro-inflammatory cytokines. It encompasses a range of clinical manifestations such as noninfectious fever, rash, diarrhea, organ dysfunction, and features of capillary leak (noncardiogenic pulmonary edema, weight gain) with no alternative etiology other than engraftment [28]. Although no definitive biomarkers of ES have been identified, elevated levels of C-reactive protein have been associated with ES [29, 30]. The diagnosis of ES is clinical, and a standard diagnostic criterion has not been established. The Spitzer [27] and Maiolino [31] diagnostic criteria for ES have been proposed (Table 52.1).

Differences in diagnostic criteria used likely account for the wide range (7–90%) of reported incidence [32]. Because the clinical presentation of ES can be seen with infection, a comprehensive workup for infectious sources should be performed. If infection is ruled out, systemic corticosteroids can be used for treatment with favorable responses, although the majority of ES cases are mild and self-limited.

Peri-engraftment respiratory distress syndrome (PERDS) likely represents a pulmonary-predominant presentation of ES. PERDS can be defined by the presence of fever, hypoxia, and diffuse lung infiltrates on radiographs in the absence of infection, cardiac dysfunction, or fluid overload, within 5 days of neutrophil recovery [33]. Bronchoalveolar lavage

 Table 52.1
 Proposed diagnostic criteria of engraftment syndrome

	Spitzer criteria [27]	Maiolino criteria [31]
Requirements	3 major or 2 major +1 minor	Major +1 minor
Major criteria	Noninfectious fever (≥101 °F) Erythrodermatous rash over >25% of the body not linked to medication Noncardiogenic pulmonary edema	Noninfectious fever
Minor criteria	Weight gain (2.5% increase) Hepatic dysfunction (bilirubin ≥2 mg/dL or a twofold increase in transaminase over baseline) Renal dysfunction (twofold increase in serum creatinine over baseline) Transient encephalopathy	Skin rash Pulmonary infiltrates Diarrhea
Timing of symptoms relative to engraftment	96 hours within ANC 500/uL	Commencing 24 hours before or at any time after the appearance of neutrophils

ANC absolute neutrophil count

(BAL) can assist in ruling out infection and assessing for diffuse alveolar hemorrhage (DAH). DAH is characterized by the return from BAL becoming progressively bloodier or demonstrating $\geq 20\%$ iron-laden macrophages on staining [34]. When no infectious etiology is identified, the primary therapy for both PERDS and DAH is with systemic corticosteroids. However, a higher dose of steroids is recommended for DAH, given as pulse doses of methylprednisolone 1 g daily, whereas in ES or PERDS, the dose of steroids is 0.5-1 mg/kg prednisone equivalent daily.

Autologous GVHD

A GVHD-like syndrome may occur in patients following autologous HCT, termed autologous GVHD. Although there may be overlapping clinical manifestations observed in ES and autologous GVHD, both are generally considered as distinct syndromes. Autologous GVHD is supported by histologic findings of classic GVHD (apoptosis with or without lymphoid infiltrate). The pathogenesis of autologous GVHD is not well understood, but is hypothesized to be related to immune dysregulation and inappropriate recognition of selfantigens [35]. Autologous GVHD is clinically and histologically similar to traditional acute GVHD, and can involve the skin, liver, and, most commonly, the GI tract [36]. Systemic corticosteroid is the mainstay of therapy and generally effective. Mild cases of autologous GVHD involving the skin can be treated with topical steroids only.

Transfusion-Associated GVHD

Transfusion-associated GVHD (TA-GVHD) is a rare but usually fatal complication of blood-component transfusion therapy. TA-GVHD occurs when viable donor lymphocytes proliferate in susceptible patients after transfusion. Patients with primary immunodeficiency syndromes, hematologic malignancies, and following HCT are at risk for TA-GVHD. The clinical features of TA-GVHD are similar to those of traditional GVHD but with an earlier onset, generally occurring 7–10 days after transfusion [35]. Patients may present with rash, fever, transaminitis, diarrhea, or abdominal pain [37]. One distinguishing clinical feature of TA-GVHD is profound pancytopenia; this increases the risk of infections, the most common cause of death in TA-GVHD, often occurring 3-4 weeks after the diagnosis [35]. Given the lack of effective therapy for TA-GVHD and its high fatality rate (80-90%), prevention of TA-GVHD is critical [38]. Irradiation of whole blood and cellular components is the widely adopted technique only to prevent TA-GVHD. Irradiation inactivates donor T lymphocytes, thereby abrogating the potential for GVHD. The US Food

and Drug Administration recommends irradiation at a dose of 2500 cGy [38]. In addition to using irradiated blood products, patients with hematologic malignancy and following HCT should also only receive blood products that are leukoreduced to avoid alloimmunization and cytomegalovirus (CMV) transmission.

Acute Complications After Allogeneic HCT

Acute toxicities related to high-dose chemotherapy used as part of conditioning prior to autologous or allogeneic HCT are similar. However, allogeneic HCT is associated with distinct acute complications not present after autologous HCT due to the presence of donor cell alloreactivity (graft-versushost disease (GVHD)), immunosuppressive medications used to prevent or treat GVHD and prevent graft rejection, and the resulting greater degree of host immunosuppression. Here we discuss acute complications unique to allogeneic HCT recipients.

Acute Infectious Complications After Allogeneic HCT

In recipients of HCT, the risk of infection is greatest during the period of pancytopenia following conditioning chemotherapy and/or radiation, but persists for 6–12 months after autologous HCT and up to 24 months following allogeneic HCT, due to the need for B- and T-cell immune reconstitution. In the allogeneic transplant recipient, this may be exacerbated by ongoing need for immunosuppression for prevention or treatment of graft-versus-host disease (GVHD). Prior to engraftment after HCT, mucosal and cutaneous damage from the conditioning regimen presents the main risk for infection. Neutropenic fever is common during this period and occurs in most patients. In the early period (<100 days after HCT), a variety of infections can occur, as summarized in Table 52.2. Specific pathogens which are important to consider in the HCT recipient are also outlined in Table 52.2.

Veno-Occlusive Disease (VOD) (Previously Termed Sinusoidal Obstruction Syndrome)

Liver complications after allogeneic HCT have become less frequent over time due to better understanding regarding the prevention and treatment of serious hepatobiliary problems. The use of ursodeoxycholic acid, a hydrophilic bile acid, during the peri-transplant period has significantly contributed to improved outcomes after HCT by reducing hepatic complications and severe acute GVHD, thus improving survival [39]. VOD is a clinical syndrome characterized by ten-

Table 52.2 Acute infections in the early post-HCT patient

Sign/		
symptom	Pathogens	Differential diagnosis
Diarrhea	CMV colitis <i>Clostridium difficile</i> Adenovirus Enteric organisms (coxsackie A, rotavirus, norovirus, echovirus) <i>Cryptosporidium</i> <i>parvum</i> (rare)	Acute GVHD Medication side effect Mucosal injury due to conditioning chemotherapy/radiation
Pneumonia	Invasive aspergillosis Mucormycosis <i>Fusarium</i> species CMV Respiratory viruses Pneumocystis pneumonia Disseminated strongyloidiasis (rare) Disseminated toxoplasmosis (rare)	Pulmonary edema Engraftment syndrome Diffuse alveolar hemorrhage IPS Drug-/radiation-induced pneumonitis Aspiration
Hepatitis	Hepatitis viruses (HBV, HCV most common) CMV EBV HHV-6 Adenovirus Hepatosplenic candidiasis	Medication toxicity Acute GVHD Iron overload Sinusoidal obstruction syndrome
Encephalitis	HHV-6 HSV VZV CMV EBV JC polyomavirus West Nile virus Adenovirus Toxoplasmosis	Medications PRES
Hemorrhagic cystitis	BK polyomavirus (common) Adenovirus (less frequent) CMV, JC polyomavirus (rare)	Drug-induced cystitis (cyclophosphamide)

HCT hematopoietic cell transplantation, *CMV* cytomegalovirus, *GVHD* graft-versus-host disease, *IPS* idiopathic pneumonia syndrome, *EBV* Epstein-Barr virus, *HHV* human herpesvirus, *HSV* herpes simplex virus, *VZV*, varicella-zoster virus, *JC*, John Cunningham virus

der hepatomegaly, fluid retention/weight gain, and elevated serum bilirubin in patients who receive high-dose myeloablative conditioning chemotherapy and/or radiation. Patients receiving high-dose cyclophosphamide (120 mg/kg) and total body irradiation (TBI) over 14 Gy are at greatest risk. In addition, pre-existing liver disease poses a significant increased hazard of VOD. The clinical presentation includes enlarged liver size, right upper quadrant tenderness, weight gain, and renal sodium retention occurring 10–20 days after the start of a myeloablative conditioning regimen. Elevated bilirubin occurs 4–10 days after these preceding signs, and concurrent portal hypertension, thrombocytopenia, and renal and lung dysfunction usually accompany the overall presentation. Diagnosis is based on clinical presentation, although liver imaging and transvenous liver biopsy and/or measurement of hepatic venous pressure may provide additional guidance [40]. Treatment of VOD is largely supportive, including management of fluid and sodium balance and maximizing renal blood flow. Patients with multi-organ failure may require hemodialysis or mechanical ventilation. A phase III trial of defibrotide for treatment of severe VOD with multi-organ failure showed improvement in day 100 survival and response; however, outcomes in patients given defibrotide were compared to historical controls. The day 100 survival was 38% with defibrotide vs. 25% in controls (p = 0.01) [41].

Acute Pulmonary Complications After Allogeneic HCT

Respiratory failure is the most common reason for intensive care unit (ICU) admissions among HCT recipients, affecting approximately 60% of HCT recipients requiring ICU care [42]. Differential diagnoses of hypercapnic and hypoxemic respiratory failure in the HCT recipient are outlined in Table 52.3.

Idiopathic pneumonia syndrome (IPS) is a term used to describe a clinical syndrome including a spectrum of noninfectious diffuse lung injuries following HCT. Diagnosis requires evidence of widespread alveolar injury, absence of active lower respiratory tract infection, and absence of a cardiac, renal, or iatrogenic etiology [43, 44]. The incidence of IPS is reported to be 4–12% associated with high case fatality rates (60–86%) in the first 100–120 days fol-

 Table 52.3
 Differential diagnoses for acute respiratory failure in the HCT recipient

Hypercapneic	T
respiratory failure	Hypoxemic respiratory failure
Neurologic causes	Infectious pneumonia
Opiates and sedative	Pulmonary edema from volume overload
medications	Congestive heart failure
Intracranial	Pneumonitis caused by aspiration, infection,
hemorrhage	chemotherapy, or other medications
PRES (see below)	Idiopathic pneumonia syndrome (includes
Airway obstruction	peri-engraftment respiratory distress
Underlying	syndrome, acute fibrinous organizing
conditions (COPD,	pneumonia, delayed pulmonary toxicity
asthma)	syndrome, diffuse alveolar hemorrhage)
Laryngeal edema due	Pulmonary embolism
to mucositis	Pneumothorax
Early-onset acute	Cryptogenic organizing pneumonia
bronchiolitis	
obliterans syndrome	
(BOS)	

PRES posterior reversible encephalopathy syndrome

lowing HCT [43, 45]. Pathological specimens will show diffuse alveolar damage and interstitial pneumonitis. Treatment includes high-dose steroids and use of TNF-alpha inhibitors such as etanercept. IPS is thought to be triggered by damage due to conditioning chemotherapy and radiation leading to endothelial injury and oxidative stress, propagated by host innate and donor T-cell immune reactions [46].

In a subset of patients with IPS, pulmonary hemorrhage or hemorrhagic alveolitis occurs. Diffuse alveolar hemorrhage (DAH) occurs in 5-12% of HCT recipients at a median time to onset of 12-19 days following both autologous and allogeneic HCTs [44]. This develops secondary due to diffuse alveolar damage from high-dose conditioning chemotherapy and/or radiation, exacerbated by thrombocytopenia. Infection may trigger this presentation as well. While radiographic features are often telling, the diagnosis is confirmed bv bronchoalveolar lavage showing increasingly bloody return of lavage fluid. Treatment involves high-dose steroids (up to 2 mg/kg/day), platelet transfusions (to keep a threshold of at least 50,000), and supportive care. DAH is associated with high mortality rates, up to 60–100% [44]. A slow and prolonged steroid taper is advised following initial response to therapy.

Acute Renal Complications After Allogeneic HCT

Evaluation of the post-HCT patient with acute elevation in serum creatinine level requires a vigilant search for the underlying cause. Workup should include complete urinalysis and urine albumin/creatinine ratio, as well as workup for hemolysis, including complete blood count with blood smear, lactate dehydrogenase, haptoglobin, and serum level of calcineurin inhibitor (if applicable). Opportunistic viral infections such as BK virus and adenovirus may impact renal function or cause viral nephropathy and should be tested in the serum. Kidney ultrasound should also be considered where clinically indicated.

Transplantation-Associated Thrombotic Microangiopathy (TMA)

The presence of renal dysfunction, microangiopathic hemolytic anemia, elevated lactate dehydrogenase, and negative workup for immune-mediated hemolysis (negative direct and indirect Coombs tests), associated with neurologic dysfunction points toward a clinical diagnosis of TMA [47]. Guidelines propose that concurrent renal and neurologic dysfunction and defined schistocyte count are not required for TMA diagnosis [48]. There is no defined or effective treatment. Discontinuation of the use of calcineurin inhibitors and replacement with other methods of GVHD prophylaxis, such as sirolimus or mycophenolate mofetil, can be considered but are not proven strategies. Posttransplant considerations such as graft-versus-host disease (described below) may limit options for a switch in immunosuppressive treatment. Other therapies that have been investigated include plasma exchange with or without rituximab. Patients with elevation in plasma levels of C5b-9 may benefit from therapy with eculizumab, although clinical trials have shown mixed results [49, 50].

Acute Neurologic Complications After Allogeneic HCT

Chemotherapy used as part of the conditioning regimen prior to infusion of donor stem cells, as well as the immunosuppressive medications used for prevention of GVHD, is associated with a variety of neurologic complications. Among these, calcineurin inhibitors (cyclosporine and tacrolimus) are most commonly used and are associated with tremors, seizures, and posterior reversible encephalopathy syndrome (PRES). PRES is characterized by rapidly evolving (over hours to days) symptoms, including headache, altered consciousness, visual disturbances, and seizures [51]. Elevated serum levels of immunosuppressive medications are not related to the onset of neurologic symptoms [52]. Less common side effects of tacrolimus include brachial plexopathy, optic neuropathy, and hearing loss; cyclosporine has been associated with pseudotumor cerebri [53]. Other possible neurologic complications of chemotherapy agents commonly used in conditioning for allogeneic HCT are summarized in Table 52.4. In addition, prophylactic antiviral medications (e.g., acyclovir) and antibiotics frequently used in this population (e.g., cefepime, imipenem) are associated with risk of seizure. Cefepime and acyclovir may cause encephalopathy as well. Voriconazole commonly causes visual hallucinations in patients who take this for prophylaxis or treatment of invasive fungal infection in the peritransplant period.

 Table 52.4
 Neurologic complications of drugs commonly used in allogeneic HCT conditioning regimens

Drug	Neurologic complication
Busulfan	Seizure
Melphalan	
Cytarabine	
Cyclophosphamide	Posterior reversible encephalopathy syndrome
Fludarabine	
Melphalan	
Cytarabine	Cerebellar dysfunction
Cytarabine	Lymphocytic meningitis
Thiotepa	
Alemtuzumab	PML (progressive multifocal
Rituximab	leukoencephalopathy)
Fludarabine	

In addition to drug-induced neurologic complications, following conditioning pancytopenia chemotherapy increases the risk of intracranial bleeding, such as subdural hematoma and intraparenchymal hemorrhage. Neutropenia and immune compromise lead to increased risk of opportunistic CNS infections, including viral encephalitis from herpes simplex, Epstein-Barr virus, varicella-zoster virus, CMV, and HHV-6. Other etiologies may include fungal (Aspergillus, Rhizopus, Candida spp.) and parasitic organisms (Toxoplasma gondii).

Acute GVHD

Acute graft-versus-host disease (aGVHD) is a frequent and potentially life-threatening complication that occurs within the first 3–4 months following allogeneic HCT. It is an inflammatory process mediated by immune-competent T cells contained in the donor graft that recognize tissues of the recipient as "foreign." The GI tract, liver, and skin are the primary organs involved. Despite prophylaxis with immunosuppressive agents, aGVHD requiring treatment occurs in approximately 50% of patients after an HLA-matched related donor HCT. The incidence and severity of aGVHD increase with a higher degree of HLA disparity between the donor and recipient. Manifestations of aGVHD are outlined in Table 52.5.

The diagnosis of acute GVHD is made clinically based on classic signs and symptoms occurring within the first several weeks to months of allogeneic HCT. However, the diagnosis is not always straightforward, and alternative diagnoses (e.g., drugs, infection, residual toxicities from preparative regimen, VOD) must be ruled out. Biopsy of the involved tissues (i.e., skin and/or GI mucosa) helps corroborate the diagnosis of GVHD. Due to the increase risk of bleeding, especially in

 Table 52.5
 Clinical manifestations of acute graft-versus-host disease (GVHD)

Organ	
involvement	Clinical manifestation
Skin	Erythematous maculopapular rash, often initially involving the palms and soles. May progress to involve the entire body surface and may be pruritic and painful in severe cases. Bullae may form leading to desquamation
Liver	Cholestasis with or without frank jaundice. Cholestasis enzymes are comparatively more elevated than transaminases
Gastrointestinal	Upper tract: anorexia, nausea, and vomiting Lower tract: diarrhea. In severe cases, may contain blood and mucosa and may be accompanied by abdominal cramping and paralytic ileus Endoscopy: in mild cases, may only show edema and erythema. In more severe cases, may show hemorrhage and ulcerations

 Table 52.6
 Clinical organ staging and grading of acute graft-versushost disease (GVHD)

			Gastrointestinal
Stage	Skin	Liver	(quantity of diarrhea)
1	Maculopapular rash <25% of body area	Bilirubin 2.0–3.0 mg/ dL	500–1000 mL
2	Maculopapular rash 25–50% of body area	Bilirubin 3.1–6.0 mg/ dL	1001–1500 mL
3	Generalized erythroderma	Bilirubin 6.1–15.0 mg/ dL	>1500 mL
4	Generalized erythroderma with bullae and desquamation	Bilirubin >15.0 mg/dL	Severe abdominal pain with and without ileus
Grade	Organ involvement		
Ι	Stage 1 or 2 skin involve	ement; no liver	or gut involvement
II	Stage 1 to 3 skin involve involvement	ement; grade 1	liver or gut
III	Stage 2 or 3 liver or gut	involvement	
IV	Stage 4 skin liver or gut	involvement	

the setting of expected posttransplant thrombocytopenia, liver biopsy is reserved for patients in whom the results would unequivocally affect clinical decision-making. The severity of aGVHD is determined by an assessment of the degree of involvement of the skin, liver, and GI tract (Table 52.6). This grading system should not be confused with histological grades sometimes noted in pathology reports from skin and GI mucosal biopsies.

aGVHD limited to the skin and involving less than 50% of body surface area can be easily managed with topical glucocorticoids alone. In more advanced cases, first-line treatment consists of systemic glucocorticoids while continuing the original immunosuppressive prophylaxis regimen. While a prednisone-equivalent dose of 2 mg/kg/day used to be the conventional starting dose for patients requiring systemic immunosuppressive therapy, a randomized trial has shown that lower initial doses (0.5-1.0 mg/kg/day) are safe and effective for patients who present with mild-to-moderate aGVHD [54]. Minimally absorbed oral glucocorticoids (beclomethasone dipropionate and budesonide) are typically used concurrently for GI aGVHD. There is no general consensus on treatment duration and taper rates once symptoms have resolved. We typically treat with systemic glucocorticoids for 7-10 days and then attempt a taper over 1-2 months, as tolerated.

Only ~50% of patients will have a long-term response to systemic glucocorticoids. Secondary immunosuppressive therapy should be considered for patients receiving systemic glucocorticoids who have (*i*) progressive symptoms in any organ after 3 days of treatment, (*ii*) persistent grade III symptoms after 1 week of treatment, or (*iii*) persistent grade II symptoms after 2 weeks of treatment [55]. Patients with

aGVHD resistant to treatment with glucocorticoids have a dismal long-term prognosis, with an overall survival likelihood of approximately 50% at 6 months. The oral Janus-activated kinase (JAK) 1/2 inhibitor, ruxolitinib, is only one FDA-approved agent for treatment of steroid-refractory aGVHD [56].

Patients receiving prolonged systemic immunosuppressive therapy are at an increased risk for infections; therefore, prophylaxis against fungal infections, HSV/VZV reactivation, and PJP should be instituted concurrently. In addition, weekly screening for CMV should be performed to detect early reactivation necessitating preemptive therapy. Patients with acute gut GVHD are at high risk for malnutrition. Maintaining even minimal oral caloric intake appears to have GI-protective effects and should be encouraged. If parenteral nutritional is needed, feeding solutions should be low in fiber, fat, and lactose. Complete bowel rest and TPN may be indicated, at least temporarily, in severe cases of gut GVHD. Supportive measures with antidiarrheal agents, including loperamide or octreotide, may be considered.

Graft Failure

Graft failure (GF) is defined as the lack of engraftment following autologous or allogeneic HCT. GF is classified as primary when there is no evidence of engraftment within the first month after transplant, or secondary when there is a loss of a previously functioning graft. The incidence of GF is less than 3–5% in the autologous and matched allogeneic HCT but up to 10% in haploidentical or cord blood recipients. GF is associated with a dismal prognosis with an overall survival of only 20% at 3–5 years following its diagnosis [57]. Infections and bleeding are the primary causes of mortality. Table 52.7 lists risk factors for GF.

There is currently no standard approach to the management of GF. Common management strategies include ensuring therapeutic doses of immunosuppression, discontinuation of myelosuppressive drugs, administration of growth factors and thrombopoietin analogues (eltrombopag, romiplostim),

Table 5	2.7 Ri	sk facto	rs of gra	aft failure
---------	---------------	----------	-----------	-------------

Pretransplant	Peri-transplant	Posttransplant
HLA mismatch	Graft sources (cord	Viral
HLA antibodies	blood/bone marrow stem	infections
ABO mismatch	cell source, and	GVHD
Iron overload	haploidentical donor)	Drug toxicity
Transfusion history	T-cell depletion graft	
Underlying diseases	manipulation	
(nonmalignant disease,	Reduced-intensity	
myelofibrosis, advanced	conditioning	
disease)	Low CD34+ cell count	
Splenomegaly		

HLA human leukocyte antigen, GVHD graft-versus-host disease

donor lymphocyte infusion, and CD34+ cell boost. A second HCT is the only potential curative option for patients with GF [58, 59]. There is no definitive data guiding the choice of conditioning regimen and donor type. A nonmyeloablative or reduced-intensity conditioning regimen is typically used with the second HCT to avoid unacceptable cumulative toxicities. In patients with an immune-mediated graft rejection, the use of an alternative donor, whenever possible, is recommended [59].

References

- Bacigalupo A, Ballen K, Rizzo D, Giralt S, Lazarus H, Ho V, et al. Defining the intensity of conditioning regimens: working definitions. Biol Blood Marrow Transplant. 2009;15(12):1628–33.
- CIBMTR. Center for International Blood & Marrow Transplant Research. The Medical College of Wisconsin and the National Marrow Donor Program. https://www.cibmtr.org. Accessed 6 Oct 2020.
- Smith TJ, Bohlke K, Lyman GH, Carson KR, Crawford J, Cross SJ, et al. Recommendations for the Use of WBC Growth Factors: American Society of Clinical Oncology Clinical Practice Guideline Update. J Clin Oncol. 2015;33(28):3199–212.
- Allen ES, Conry-Cantilena C. Mobilization and collection of cells in the hematologic compartment for cellular therapies: stem cell collection with G-CSF/plerixafor, collecting lymphocytes/monocytes. Semin Hematol. 2019;56(4):248–56.
- Tigue CC, McKoy JM, Evens AM, Trifilio SM, Tallman MS, Bennett CL. Granulocyte-colony stimulating factor administration to healthy individuals and persons with chronic neutropenia or cancer: an overview of safety considerations from the Research on Adverse Drug Events and Reports project. Bone Marrow Transplant. 2007;40(3):185–92.
- Stiff PJ, Bensinger W, Abidi MH, Gingrich R, Artz AS, Nademanee A, et al. Clinical and ultrasonic evaluation of spleen size during peripheral blood progenitor cell mobilization by filgrastim: results of an open-label trial in normal donors. Biol Blood Marrow Transplant. 2009;15(7):827–34.
- Hill JM, Syed MA, Arai AE, Powell TM, Paul JD, Zalos G, et al. Outcomes and risks of granulocyte colony-stimulating factor in patients with coronary artery disease. J Am Coll Cardiol. 2005;46(9):1643–8.
- Kaplan A. Complications of apheresis. Semin Dial. 2012;25(2):152–8.
- Syme R, Bewick M, Stewart D, Porter K, Chadderton T, Glück S. The role of depletion of dimethyl sulfoxide before autografting: on hematologic recovery, side effects, and toxicity. Biol Blood Marrow Transplant. 2004;10(2):135–41.
- Higman MA, Port JD, Beauchamp NJ Jr, Chen AR. Reversible leukoencephalopathy associated with re-infusion of DMSO preserved stem cells. Bone Marrow Transplant. 2000;26(7):797–800.
- Chen-Plotkin AS, Vossel KA, Samuels MA, Chen MH. Encephalopathy, stroke, and myocardial infarction with DMSO use in stem cell transplantation. Neurology. 2007;68(11):859–61.
- Windrum P, Morris TC, Drake MB, Niederwieser D, Ruutu T. Variation in dimethyl sulfoxide use in stem cell transplantation: a survey of EBMT centres. Bone Marrow Transplant. 2005;36(7):601–3.
- Schubert MM, Pizzigatti Correa ME, Peterson DE. Oral complications of hematopoietic cell transplantation. In: Forman SJ, Negrin RS, Antin HJ, Applebaum FR, editors. Thomas' hematopoietic

cell transplantation: stem cell transplantation. Oxford, UK: Wiley; 2016. p. 1242–56.

- Spielberger R, Stiff P, Bensinger W, Gentile T, Weisdorf D, Kewalramani T, et al. Palifermin for oral mucositis after intensive therapy for hematologic cancers. N Engl J Med. 2004;351(25):2590–8.
- Hensley ML, Hagerty KL, Kewalramani T, Green DM, Meropol NJ, Wasserman TH, et al. American Society of Clinical Oncology 2008 Clinical Practice Guideline Update: use of chemotherapy and radiation therapy protectants. J Clin Oncol. 2009;27(1):127–45.
- 16. Lilleby K, Garcia P, Gooley T, McDonnnell P, Taber R, Holmberg L, et al. A prospective, randomized study of cryotherapy during administration of high-dose melphalan to decrease the severity and duration of oral mucositis in patients with multiple myeloma undergoing autologous peripheral blood stem cell transplantation. Bone Marrow Transplant. 2006;37(11):1031–5.
- Feyer P, Hoffmann FA, Standke E, Kubel M, Helbig W. Acute and late effects in the course of and after total body irradiation with following bone marrow transplantation in patients with different forms of leukaemia. Folia Haematol Int Mag Klin Morphol Blutforsch. 1989;116(3–4):487–91.
- Crump M, Brandwein JM, Scott JG, Sutcliffe SB, Keating A. Acute transient parotitis after high dose etoposide and autologous bone marrow transplantation. Bone Marrow Transplant. 1990;6(4):259–61.
- McDonald LC, Gerding DN, Johnson S, Bakken JS, Carroll KC, Coffin SE, et al. Clinical Practice Guidelines for *Clostridium difficile* Infection in Adults and Children: 2017 Update by the Infectious Diseases Society of America (IDSA) and Society for Healthcare Epidemiology of America (SHEA). Clin Infect Dis. 2018;66(7):987–94.
- Jain T, Croswell C, Urday-Cornejo V, Awali R, Cutright J, Salimnia H, et al. *Clostridium difficile* colonization in hematopoietic stem cell transplant recipients: a prospective study of the epidemiology and outcomes involving toxigenic and nontoxigenic strains. Biol Blood Marrow Transplant. 2016;22(1):157–63.
- Koo HL, Koo DC, Musher DM, DuPont HL. Antimotility agents for the treatment of *Clostridium difficile* diarrhea and colitis. Clin Infect Dis. 2009;48(5):598–605.
- Wilcox MH, Howe R. Diarrhoea caused by Clostridium difficile: response time for treatment with metronidazole and vancomycin. J Antimicrob Chemother. 1995;36(4):673–9.
- Hockenbery DM, Strasser SI, McDonald GB. Gastrointestinal and hepatic complications. In: Forman SJ, Negrin RS, Antin HJ, Applebaum FR, editors. Thomas' hematopoietic cell transplantation: stem cell transplantation. Oxford, UK: Wiley; 2016. p. 1140–55.
- 24. Urbach DR, Rotstein OD. Typhlitis. Can J Surg. 1999;42(6):415-9.
- Kirkpatrick ID, Greenberg HM. Gastrointestinal complications in the neutropenic patient: characterization and differentiation with abdominal CT. Radiology. 2003;226(3):668–74.
- 26. Taplitz RA, Kennedy EB, Bow EJ, Crews J, Gleason C, Hawley DK, et al. Outpatient Management of Fever and Neutropenia in Adults Treated for Malignancy: American Society of Clinical Oncology and Infectious Diseases Society of America Clinical Practice Guideline Update. J Clin Oncol. 2018;36(14):1443–53.
- 27. Spitzer TR. Engraftment syndrome following hematopoietic stem cell transplantation. Bone Marrow Transplant. 2001;27(9):893–8.
- Cornell RF, Hari P, Drobyski WR. Engraftment syndrome after autologous stem cell transplantation: an update unifying the definition and management approach. Biol Blood Marrow Transplant. 2015;21(12):2061–8.
- 29. Fassas AB, Miceli MH, Grazzlutti M, Dong L, Barlogie B, Anaissie E. Serial measurement of serum C-reactive protein levels can identify patients at risk for severe complications following autologous stem cell transplantation. Leuk Lymphoma. 2005;46(8):1159–61.

- 30. Carreras E, Fernández-Avilés F, Silva L, Guerrero M, de Larrea CF, Martínez C, et al. Engraftment syndrome after auto-SCT: analysis of diagnostic criteria and risk factors in a large series from a single center. Bone Marrow Transplant. 2010;45(9):1417–22.
- Maiolino A, Biasoli I, Lima J, Portugal AC, Pulcheri W, Nucci M. Engraftment syndrome following autologous hematopoietic stem cell transplantation: definition of diagnostic criteria. Bone Marrow Transplant. 2003;31(5):393–7.
- Spitzer TR. Engraftment syndrome: double-edged sword of hematopoietic cell transplants. Bone Marrow Transplant. 2015;50(4):469–75.
- 33. Capizzi SA, Kumar S, Huneke NE, Gertz MA, Inwards DJ, Litzow MR, et al. Peri-engraftment respiratory distress syndrome during autologous hematopoietic stem cell transplantation. Bone Marrow Transplant. 2001;27(12):1299–303.
- Afessa B, Tefferi A, Litzow MR, Krowka MJ, Wylam ME, Peters SG. Diffuse alveolar hemorrhage in hematopoietic stem cell transplant recipients. Am J Respir Crit Care Med. 2002;166(5):641–5.
- 35. Cutler C, Antin JH. Manifestations and treatment of acute graftversus-host disease. In: Forman SJ, Negrin RS, Antin HJ, Applebaum FR, editors. Thomas' hematopoietic cell transplantation: stem cell transplantation. Oxford, UK: Wiley; 2016. p. 1012–25.
- 36. Holmberg L, Kikuchi K, Gooley TA, Adams KM, Hockenbery DM, Flowers ME, et al. Gastrointestinal graft-versus-host disease in recipients of autologous hematopoietic stem cells: incidence, risk factors, and outcome. Biol Blood Marrow Transplant. 2006;12(2):226–34.
- Kopolovic I, Ostro J, Tsubota H, Lin Y, Cserti-Gazdewich CM, Messner HA, et al. A systematic review of transfusion-associated graft-versus-host disease. Blood. 2015;126(3):406–14.
- Moroff G, Luban NL. The irradiation of blood and blood components to prevent graft-versus-host disease: technical issues and guidelines. Transfus Med Rev. 1997;11(1):15–26.
- Ruutu T, Eriksson B, Remes K, Juvonen E, Volin L, Remberger M, et al. Ursodeoxycholic acid for the prevention of hepatic complications in allogeneic stem cell transplantation. Blood. 2002;100(6):1977–83.
- McDonald GB. Hepatobiliary complications of hematopoietic cell transplantation, 40 years on. Hepatology. 2010;51(4):1450–60.
- 41. Richardson PG, Riches ML, Kernan NA, Brochstein JA, Mineishi S, Termuhlen AM, et al. Phase 3 trial of defibrotide for the treatment of severe veno-occlusive disease and multi-organ failure. Blood. 2016;127(13):1656–65.
- 42. Afessa B, Azoulay E. Critical care of the hematopoietic stem cell transplant recipient. Crit Care Clin. 2010;26(1):133–50.
- 43. Wenger DS, Triplette M, Crothers K, Cheng GS, Hill JA, Milano F, et al. Incidence, risk factors, and outcomes of idiopathic pneumonia syndrome after allogeneic hematopoietic cell transplantation. Biol Blood Marrow Transplant. 2020;26(2):413–20.
- 44. Panoskaltsis-Mortari A, Griese M, Madtes DK, Belperio JA, Haddad IY, Folz RJ, et al. An official American Thoracic Society research statement: noninfectious lung injury after hematopoietic stem cell transplantation: idiopathic pneumonia syndrome. Am J Respir Crit Care Med. 2011;183(9):1262–79.
- 45. Fukuda T, Hackman RC, Guthrie KA, Sandmaier BM, Boeckh M, Maris MB, et al. Risks and outcomes of idiopathic pneumonia syndrome after nonmyeloablative and conventional conditioning regimens for allogeneic hematopoietic stem cell transplantation. Blood. 2003;102(8):2777–85.
- Vande Vusse LK, Madtes DK. Early onset noninfectious pulmonary syndromes after hematopoietic cell transplantation. Clin Chest Med. 2017;38(2):233–48.
- Hingorani S. Renal complications of hematopoietic-cell transplantation. N Engl J Med. 2016;374(23):2256–67.
- Cho BS, Yahng SA, Lee SE, Eom KS, Kim YJ, Kim HJ, et al. Validation of recently proposed consensus criteria for thrombotic

microangiopathy after allogeneic hematopoietic stem-cell transplantation. Transplantation. 2010;90(8):918–26.

- 49. de Fontbrune FS, Galambrun C, Sirvent A, Huynh A, Faguer S, Nguyen S, et al. Use of eculizumab in patients with allogeneic stem cell transplant-associated thrombotic microangiopathy: a study from the SFGM-TC. Transplantation. 2015;99(9):1953–9.
- Jodele S, Fukuda T, Vinks A, Mizuno K, Laskin BL, Goebel J, et al. Eculizumab therapy in children with severe hematopoietic stem cell transplantation-associated thrombotic microangiopathy. Biol Blood Marrow Transplant. 2014;20(4):518–25.
- Fugate JE, Claassen DO, Cloft HJ, Kallmes DF, Kozak OS, Rabinstein AA. Posterior reversible encephalopathy syndrome: associated clinical and radiologic findings. Mayo Clin Proc. 2010;85(5):427–32.
- 52. Schwartz RB, Bravo SM, Klufas RA, Hsu L, Barnes PD, Robson CD, et al. Cyclosporine neurotoxicity and its relationship to hypertensive encephalopathy: CT and MR findings in 16 cases. AJR Am J Roentgenol. 1995;165(3):627–31.
- Dulamea AO, Lupescu IG. Neurological complications of hematopoietic cell transplantation in children and adults. Neural Regen Res. 2018;13(6):945–54.
- 54. Mielcarek M, Furlong T, Storer BE, Green ML, McDonald GB, Carpenter PA, et al. Effectiveness and safety of lower dose prednisone for initial treatment of acute graft-versus-host disease: a randomized controlled trial. Haematologica. 2015;100(6):842–8.

- 55. Martin PJ, Rizzo JD, Wingard JR, Ballen K, Curtin PT, Cutler C, et al. First- and second-line systemic treatment of acute graftversus-host disease: recommendations of the American Society of Blood and Marrow Transplantation. Biol Blood Marrow Transplant. 2012;18(8):1150–63.
- 56. Jagasia M, Perales MA, Schroeder MA, Ali H, Shah NN, Chen YB, et al. Ruxolitinib for the treatment of steroid-refractory acute GVHD (REACH1): a multicenter, open-label phase 2 trial. Blood. 2020;135(20):1739–49.
- 57. Olsson R, Remberger M, Schaffer M, Berggren DM, Svahn BM, Mattsson J, Ringden O. Graft failure in the modern era of allogeneic hematopoietic SCT. Bone Marrow Transplant. 2013;48(4):537–43. Erratum in: Bone Marrow Transplant. 2013;48(4):616.
- 58. Cesaro S, Peffault de Latour R, Tridello G, Pillon M, Carlson K, Fagioli F, et al; Severe Aplastic Anaemia Working Party of the European Group for Blood and Marrow Transplantation. Second allogeneic stem cell transplant for aplastic anaemia: a retrospective study by the Severe Aplastic Anaemia Working Party of the European Society for Blood and Marrow Transplantation. Br J Haematol. 2015;171(4):606–14.
- Locatelli FLB, Merli P. Current and future approaches to treat graft failure after allogeneic hematopoietic stem cell transplantation. Expert Opin Pharmacother. 2014;15(1):23–36.



Toxicities of Novel Antineoplastic Therapies

53

Eva Rajha and Sai-Ching Jim Yeung

Introduction

Many novel cancer therapy agents, altogether about a hundred, have been approved by the US Food and Drug Administration (FDA) between 2011 and 2020 for hematologic malignancies (Table 53.1) and solid tumors (Table 53.2). These new treatments offer improved clinical outcomes and survival of cancer patients at the price of treatment-related adverse events. Thus, the emergence of new treatmentrelated adverse events and syndromes will present diagnostic and management challenges to emergency physicians when they provide care for the cancer patient.

Case Study

A 65-year-old man with metastatic melanoma presents to the emergency department (ED) with the chief complaints of general weakness and confusion. He is not known to have brain metastasis, and his magnetic resonance imaging (MRI) of the brain did not show any metastatic disease 1 month prior to ED presentation. For his melanoma, he is not on cytotoxic chemotherapy and received his fourth dose of immunotherapy with ipilimumab 1 week ago. He complains of progressive general weakness, and he felt too weak to get out of bed without assistance for the past 2 days. He has lost his appetite and has nausea for about 3 days, resulting in decreased oral intake. He is lethargic and disoriented but without focal neurological complaints. His review of systems reveals symptoms of mild headache and constipation.

E. Rajha

His vital signs are normal except for orthostatic hypotension. His physician examination is within normal limits except for disorientation to time, lack of concentration, inability to perform serial-7 subtraction, and lethargy. Laboratory testing reveals the following: serum sodium 128 mEq/L, blood urea nitrogen 35 mg/dL, creatinine 1.1 mg/ dL, and glucose 68 mg/dL. The rest of the comprehensive metabolic panel and complete blood count with differential are normal.

Further testing in the ED shows that cortisol level is $2.7 \mu g/dL$, thyrotropin 0.06 mIU/L, and free thyroxine 0.6 ng/dL. MRI of the sella reveals a slightly enlarged pituitary gland (1.2 cm in height, 1.1 cm in anterior-posterior dimension, and 1 cm in width). The dome of the mass is slightly indenting the inferior aspect of the optic chiasm.

The patient is diagnosed with adrenal insufficiency and central hypothyroidism due to ipilimumab-induced hypophysitis. The patient is hospitalized for further evaluation and management of this immune-related adverse effect (irAE). Endocrinology service is consulted; dynamic testing with low-dose cosyntropin stimulation test and measurements of adrenotropin corroborates central hypoadrenalism. Testing of other pituitary axes reveals panhypopituitarism.

General Approach

To appropriately and promptly diagnose and manage these toxicities, emergency physicians should be familiar with the array of adverse effect of these cancer therapies. This chapter will summarize the recent therapeutic advances in the field of oncologic and provide a reference for emergency physicians dealing with adverse events that are caused by these new agents and may present as oncologic emergencies. Some of these presentations can mimic sepsis, and we propose an algorithm to guide emergency providers in their approach to these complications (Fig. 53.1), linking to existing diagnostic guidelines.

Department of Emergency Medicine, Baylor College of Medicine, Houston, TX, USA

S.-C. J. Yeung (⊠) Department of Emergency Medicine, Division of Internal Medicine, The University of Texas MD Anderson Cancer Center, Houston, TX, USA e-mail: syeung@mdanderson.org

K. H. Todd et al. (eds.), Oncologic Emergency Medicine, https://doi.org/10.1007/978-3-030-67123-5_53

	duarca avante		ZNS events Cytokine release syndrome Hypogammaglobulinemia nfections, pancytopenia	CNS events Cytokine release syndrome typogammaglobulinemia nfections, cytopenias	CNS events Sytokine release syndrome ufections, cytopenias		mmune-mediated adverse events (most common, musculoskeletal pain, infection, fatigue, cough, lyspnea, diarrhea, abdominal pain, nausea, eadache: most serious, arrhythmia, myocardial nfarction, pericardial effusion) nfusion-related reactions	mmune-mediated adverse events (most common, ash, diarrhea, cough; most serious, pneumonia, meumonitis, pyrexia, infusion reaction) ransplantation-related complications (GVHD) lepatic veno-occlusive disease		ndirect hyperbilirubinemia Differentiation syndrome	XT prolongation .eukocytosis)ifferentiation syndrome	ebrile neutropenia Aucositis, nausea, vomiting Headache 'etechiae, epistaxis Ipper respiratory tract infection	Cardiac and arterial vascular occlusive events Vrrhythmias, HF 'ancreatitis, hepatotoxicity	keversible posterior leukoencephalopathy
	Commente		Patients with ALL up to C 25 years F In	LBCL C refractory C C C LBCL LBCL LBCL LBCL LBCL LBCL LB	Relapsed or refractory C LBCL C		Mediastinal LBCL h n d d f i i i i h h h h h h h h h h h h h h h			Relapsed or refractory acute II myeloid leukemia E	Relapsed or refractory AML C	AML <i>FLT3</i> mutation- positive (<i>FLT3</i> +), advanced N mastocytosis P P	CML, Philadelphia C chromosome-positive ALL A P	R
	Indication	TILITIC COLOR	ALL	LBCL	LBCL	spoint molecule	PMBCL, cHL	cHL		AML	AML	AML	CML, ALL	
s	Machaniem of action		CD19-directed genetically modified autologous T-cell immunotherapy		CD19-directed genetically modified autologous T-cell immunotherapy	ath protein-1 (PD-1), an immune check,	PD-1 humanized mouse mAb	PD-1 human IgG4 mAb		IDH2 inhibitor	IDH1 inhibitor	FLT3 inhibitor	BCR-ABL inhibitor	
	FDA approval data	uaiv	08/30/17		10/18/17	ummed cell dea	06/13/18	05/17/16		08/01/17	07/20/18	04/28/17	12/17/12	
C 11	Rrand name	aptor-T-cells	KYMRIAH		YESCARTA	es against progre	KEYTRUDA	OPDIVO	itors	IDHIFA	TIBSOVO	RYDAPT	ICLUSIG	
	anario nama	Chimeric antigen rece	fisagenlecleucel		Axicabtagene siloleucel	Monoclonal antibodie	2embrolizumab	Nivolumab	Small molecule inhibi	Enasidenib	vosidenib	vfidostaurin	onatinib	

 Table 53.1
 Novel drugs approved after 2010 for hematologic malignancies and their associated adverse events

Myelosuppression Diarrhea, pancreatitis, hepatotoxicity Cardiovascular events	Myelosuppression Arrhythmias	Cytopenias Hypertension	Headache, diarrhea, fatigue, myalgia Neutropenia, anemia, pneumonia Cardiac adverse events	Serious hepatotoxicity Diarrhea, colitis, perforation Pneumonitis Myelosuppression, infections	Nausea, diarrhea Pyrexia Cytopenias	Hyperglycemia, hypertension, infections, neutropenia Noninfectious pneumonitis Severe cutaneous reactions	Myelosuppression, infections Thrombotic microangiopathy Thromboembolic events Renal and hepatic toxicity Pneumonitis Tumor lysis syndrome QT prolongation Reversible posterior leukoencephalopathy	Cytopenias, peripheral neuropathy Cardiac ischemia, arrhythmias Severe diarrhea	Myelosuppression, infections Hepatic toxicity Tumor lysis syndrome HF, QT prolongation	Cytopenias, vomiting, diarrhea, constipation, neuropathy, peripheral edema, back pain	Tumor lysis syndrome Myelosuppression Autoimmune hemolytic anemia	Myelosuppression, sepsis HF Herpes zoster infection
Philadelphia chromosome- positive CML				Relapsed CLL	Refractory CLL, SLL, or FL	Relapsed FL	Relapsed/refractory PTCL					
CML	MCL	MCL, CLL, WM	MCL	CLL, NHL, SLL	CLL, SLL, FL	E	MCL	MM	MM	MM	CLL, SLL	Myelofibrosis, PV
BCR-ABL inhibitor	BCR inhibitor	Irreversible inhibitor of Bruton tyrosine kinase (BTK)	Second-generation BTK inhibitor	Phosphatidylinositol 3-kinase δ inhibitor	Phosphatidylinositol 3-kinase δ and γ (Pl3K-S, $\gamma)$ inhibitor	Pan-class I phosphoinositide 3-kinase (PI3K) inhibitor	Deacetylase I and II inhibitor	Pan-deacetylase inhibitor	Proteasome inhibitor	Proteasome inhibitor	BCL2 inhibitor	JAK inhibitor
09/04/12	11/14/2019	11/13/13	10/31/17	07/23/14	24/09/18	09/14/17	07/03/14	02/23/15	07/20/12	11/20/15	4/11/2016	11/16/11
BOSULIF	BRUKINSA	MBRUVICA	CALQUENCE	ZYDELIG	COPIKTRA	aliqopa	BELEODAQ	FARYDAK	KYPROLIS	NINLARO	VENCLEXTA	IAKAFI
Bosutinib	Zanubrutinib	Ibrutinib	Acalabrutinib	Idelalisib	Duvelisib	Copanlisib	Belinostat	Panobinostat lactate	Carfilzomib	Ixazomib citrate	Venetoclax	Ruxolitinib

713

(continued)	
- -	
ŝ	
a,	
ā	
Ta	

	Adverse events	Serious and fatal encephalopathy, thiamine deficiency Anemia, thrombocytopenia Gastrointestinal, hepatic toxicity Amylase, lipase elevation	Myelosuppression Glucose intolerance		Reactivation of hepatitis B virus infection Progressive multifocal leukoencephalopathy Tumor lysis syndrome Infusion reaction Cytopenias	Febrile neutropenia, thrombocytopenia Infusion reactions	Cytopenias, diarrhea, pneumonia	Pyrexia, anemia, pneumonia, pulmonary embolism, acute renal failure, hepatotoxicity Infusion reactions Second primary malignancies	Rash, infusion reactions GVHD if allogenic transplantation within 50 days of treatment		Veno-occlusive disease (sinusoidal occlusion syndrome) Infusion reactions Thrombocytopenia, neutropenia QT prolongation	Infection, bleeding, prolonged thrombocytopenia Veno-occlusive disease, transaminitis	Progressive multifocal leukoencephalopathy Peripheral neuropathy Myelosuppression
	Comments		CML after failure of two or more TKIs		Recurrent or progressive CLL						Relapsed or refractory B-cell precursor ALL	CD33-positive AML	Anaplastic large cell or other CD30-expressing T-cell lymphomas, relapsed or refractory Hodgkin lymphoma
	Indication	Myelofibrosis	CML		CLL	CLL, FL	MM	MM	MF		ALL	AML	T-cell lymphoma, cHL
	Mechanism of action	JAK2 inhibitor	BCR-ABL1 and Mcl-1 synthesis inhibitor		CD20 human mAb	CD20 humanized mAb	CD38 human IgG1k mAb	Signaling lymphocyte activation molecule family 7 (SLAMF7) humanized mAb	CC chemokine receptor 4 (CCR4) humanized mAb defucosylated at the Fc region to enhance antibody- mediated cellular cytotoxicity		CD22 humanized mAb covalently conjugated to N-acetyl-y- calicheamicin dimethylhydrazide	CD33 humanized mAb covalently conjugated to N-acetyl-y- calicheamicin dimethylhydrazide	CD30 chimeric mAb covalently linked with cleavable linkers to 3–5 units of monomethyl auristatin E
FDA approval	date	8/16/2019	10/26/12	urface antigens	04/17/14	11/01/13	11/21/16	11/30/15	08/08/18		<i>71/71/80</i>	09/01/17	09/19/11
	Brand name	INREBIC	SYNRIBO	s against cell su	ARZERRA	GAZYVA	DARZALEX	EMPLICITI	POTELIGEO	utes	BESPONSA	MYLOTARG	ADCETRIS
	Generic name	Fedratinib hydrochloride	Omacetaxine mepesuccinate	Monoclonal antibodie.	Ofatumumab	Obinutuzumab	Daratumumab	Elotuzumab	Mogamulizumab- kpkc	Antibody-drug conjugu	Inotuzumab ozogamicin	Gemtuzumab ozogamicin	Brentuximab vedotin

	aolytic uremic syndrome illary leak syndrome		S events okine release syndrome		ous hypersensitivity sding or thrombosis hertriglyceridemia, pancreatitis	еганннопенна		elosuppression or lysis syndrome ere cutaneous reactions h defects, fetal death omboembolic events	; picumomus • coll lumahomo <i>ett</i> clossic Uedebia lumahomo
	Relapsed or refractory HCL He Ca		B-cell precursor ALL CN Cy		ALL with hypersensitivity Ser to <i>E. coli</i> -derived Bld asparaginase Hy	пу		My See Bir Th	DMDCI mimory modioctinol long
	HCL		ALL		ALL			MM	omodamii Iloo
	CD22-directed cytotoxin		CD19-/CD3-bi-specific mAb		Cytotoxicity by catalyzing deamidation of asparagine		y	Angiogenesis inhibitor, $TNF-\alpha$ inhibitor	D control of DCI 1000000
	09/13/18		13/3/14		11/18/11	and action activity	ppiastic activit	02/08/13	AII conto IV
	LUMOXITI	iger	BLINCYTO	6	ERWINAZE	and with anting	gent with antine	POMALYST	~ Administration
Immunotoxin	Moxetumomab pasudotox-tdfk	Bi-specific T-cell engo	Blinatumomab	Protein-based therap)	Asparaginase Erwinia chrysanthemi	Turning and and atoms of	Immunomoautatory a	Pomalidomide	ZDA LIS Ecod and Day

FDA US Food and Drug Administration, *ALL* acute lymphoblastic leukemia, *LBCL* large B-cell lymphoma, *PMBCL* primary mediastinal large B-cell lymphoma, *cHL* classic Hodgkin lymphoma, *GVHD* graft-versus-host disease, *AML* acute myeloid leukemia, *CNS* central nervous system, *CML* chronic myeloid leukemia, *HF* heart failure, *MCL* mantle cell lymphoma, *CLL* chronic lympho-cytic leukemia, *WM* Waldenstrom's macroglobulinemia, *NHL* non-Hodgkin lymphoma, *SLL* small lymphocytic lymphoma, *FL* follicular lymphoma, *PTCL* relapsed/refractory peripheral T-cell lymphoma, *MM* multiple myeloma, *MF* mycosis fungoides, *HCL* hairy cell lymphoma, *PV* polycythemia vera, *ILD* interstitial lung disease

 Table 53.2
 Novel drugs approved after 2010 for solid tumors and their associated adverse events

	Adverse events		Immune-mediated adverse events (most common.	musculoskeletal pain, infection, fatigue, cough,	dyspnea, diarrhea, abdominal pain, nausea, headache;	most serious, arrhythmia, myocardial infarction,	pericardial effusion)	Infusion-related reactions						Immune-mediated adverse events (most common,	rash, diarrhea, cough; most serious, pneumonia,	pneumonitis, pyrexia, infusion reaction)	Transplantation-related complications (GVHD)	Hepatic veno-occlusive disease			Immune-mediated adverse events (most common.	fatigue, rash, diarrhea, transaminitis, infection, renal	dysfunction; most serious, myositis, colitis, adrenal	insufficiency, pneumonitis, myocarditis) Infusion-related reactions	Immune-mediated adverse events (most common,	fatigue, rash, nausea; most serious, cellulitis,	hypercalcemia, hypophysitis)	Infusion-related reactions		Immune-mediated adverse events (most common,	fatigue, rash, emesis, elevated lipase; most serious,	pneumonitis, hepatotoxicity, adrenal insufficiency)	Infusion-related reactions	Immune-mediated adverse events (most common,	fatigue, pruritus; most serious, hepatotoxicity)	Infusion-related reactions		Immune-mediated adverse events (most common,	fatigue, rash; most serious, colitis, HLH)	Infusion-related reactions
	Comments								HCC						NSCLC, SCLC					HCC		NSCLC		SCLC													volecule			
	Indication	me checkpoint molecule	Melanoma	NSCLC	Renal cell cancer	Head & neck cancer	Urothelial cancer	Gastric cancer	Cervical cancer	Liver cancer	Merkel cell cancer	SCLC Esophageal cancer	Endometrial cancer	Melanoma	Lung cancer	Renal cell cancer	Head & neck cancer	Urothelial cancer	Colorectal cancer	Liver cancer	Urothelial cancer	Lung cancer	Breast cancer	Lung cancer	Cutaneous squamous cell	carcinoma			une checkpoint molecule	Merkel cell carcinoma	Urothelial cancer	Renal cell cancer		Urothelial cancer	NSCLC		A4), an immune checkpoint m	Melanoma	Renal cell cancer	Colorectal cancer
	Mechanism of action	th protein-1 (PD-1), an immu	PD-1 humanized mouse	mAb										PD-1 human IgG4 mAb							PD-1 humanized IgG1	mAb			PD-1 recombinant human	IgG4 mAb			th ligand I (PD-LI), an imm	PD-L1 human IgG1 mAb				PD-L1 human IgG1κ mAb			te-associated antigen 4 (CTL	CTLA4 human mAb		
EDA	approval date	mmed cell dea	09/04/14	10/02/15	11/23/15	08/05/16	05/18/17	09/22/17	06/12/18	11/09/18	12/19/18	07/30/19	09/17/19	12/22/14	03/04/15	11/23/15	11/10/16	02/02/17	08/01/17	09/22/17	05/18/16	10/18/16	03/08/19	03/18/19	09/28/18				mmed cell dea	03/23/17	05/09/17	05/14/19		05/01/17	02/16/18		cic T lymphocy	03/25/11	16/4/18	11/6/18
	trand name	s against progra	TRUDA											OPDIVO							ECENTRIO				JBTAYO				s against progra	AVENCIO				MFINZI			s against cytoto.	(ERVOY		
	Generic name B	Monoclonal antibodie.	Pembrolizumah k											Nivolumab							Atezolizumab				Cemiplimab-rwlc L				Monoclonal antibodie.	Avelumab E				Durvalumab			Monoclonal antibodie.	Ipilimumab 1		

	QT prolongation Hypertension, HF Dermatologic toxicity Interstitial lung disease	Hypertension Serious hepatotoxicity Dermatologic toxicity Reversible posterior leukoencephalopathy	Hypertension, HF Dermatologic toxicity Thrombotic events	Hyperphosphatemia, nausea, stomatitis, dry mouth Central serous retinopathy/retinal pigment epithelial detachment	Gastrointestinal toxicity, diarrhea	Hypertension, cardiac dysfunction, QT prolongation, arterial thromboembolic events Renal and liver toxicity Hemorrhagic events Reversible posterior leukoencephalopathy Palmoplantar erythrodysesthesia	Hypertensive crisis Hemorrhage, perforations, fistulas, wound complications Thrombotic events Diarrhea Osteonecrosis of the jaw Reversible posterior leukoencephalopathy Palmoplantar erythrodysesthesia	Hepatic and renal toxicity ILD/pneumonitis Bradycardia Myalgia and CPK elevation	Myelosuppression QT prolongation, bradycardia Liver and lung toxicity Ocular toxicity	QT prolongation, bradycardia Gastrointestinal, liver, lung, and ocular toxicity Pancreatitis
				Locally advanced or metastatic urothelial carcinoma with abnormal FGFR gene		НСС		ALK-positive NSCLC	ALK-positive or ROSI-positive NSCLC	ALK-positive NSCLC
	Thyroid cancer	Colorectal cancer Advanced GIST tumor Liver cancer	Renal cell cancer	Bladder cancer	Breast cancer	Endometrial cancer Renal cell cancer Liver cancer Thyroid cancer	Thyroid cancer Renal cell cancer Liver cancer	NSCLC	NSCLC	NSCLC
	VEGFR inhibitor	VEGFR inhibitor	VEGFR inhibitor	Pan FGFR kinase inhibitor	Irreversible HER2 and EGFR inhibitor	VEGFR inhibitor	RTK inhibitor	ALK inhibitor	ALK inhibitor	ALK inhibitor
	04/06/11	10/27/12	01/27/12	04/12/19	07/17/17	02/13/15	04/25/16 04/25/16	12/11/15	11/20/13	04/29/14
bitors	VANDETANIB	STIVARGA	INLYTA	BALVERSA	NERLYNX	LENVIMA	CABOMETYX CABOMETYX	ALECENSA	XALKORI	ZYKADIA
Small molecule inhi	Vandetanib	Regorafenib	Axitinib	Erdafitinib	Neratinib maleate	Lenvatinib mesylate	Cabozantinib-s- malate	Alectinib	Crizotinib	Ceritinib

(continued)

(continued)
Ч
m
ŝ
Ð
5
Ta
- C

	Adverse events ILD/pneumonitis Hypertension, bradycardia Myalgia and CPK elevation Visual disturbance Pancreatic enzymes elevation Hyperglycemia	Serious hepatotoxicity ILD/pneumonitis Atrioventricular block CNS disturbance Hyperlipidemia	Liver and renal toxicity CHF CNS and visual disturbance Skeletal fractures QT prolongation Hyperuricemia	LLD/pneumonitis QT prolongation Cardionryopathy Diarrhea, rash	ILD/pneumonitis Diarrhea, rash	ILD/pneumonitis Cardiovascular toxicity Gastrointestinal and liver toxicity Severe cutaneous reaction	Liver and renal toxicity Pancreatitis QT prolongation Severe cutaneous reactions Cutaneous malignancies Radiation sensitization Severe hypersensitivity Dupuytren contracture, plantar fascial fibromatosis	Severe cutaneous reactions New cutaneous malignancy Hemorrhage, febrile reaction Cardiomyopathy, QT prolongation ILD/pneumonitis Ocular toxicity Venous thromboembolism
	ALK-positive NSCLC	ALK-positive or ROS1-positive NSCLC	ROS 1-positive NSCLC Solid tumors with NTRK gene fusion	EGFR T790M mutation-positive NSCLC	NSCLC with EGFR exon 19 del or exon 21 L858R substitution mutations	NSCLC with EGFR exon 19 del or exon 21 L858R substitution mutations	Melanoma with BRAF V600E mutation	Melanoma with BRAF V600E mutation
	Indication	NSCLC	NSCLC Solid tumors with NTRK gene fusion	NSCLC	NSCLC	NSCLC	Melanoma	Melanoma
	Mechanism of action ALK inhibitor	RTK, ALK, ROS1 inhibitor	Pan-TRK ROS1, ALK inhibitor	Mutant selective EGFR inhibitor	Second-generation pan EGFR inhibitor	Pan-HER and EGFR inhibitor	BRAF inhibitor	BRAF inhibitor
FDA approval	date 04/28/17	11/02/18	08/15/19	11/13/15	09/27/18	06/12/13	08/17/11	05/29/13
	Brand name ALUNBRIG	LORBRENA	ROZLYTREK	TAGRISSO	VIZIMPRO	GILOTRIF	ZELBORAF	TAFINLAR
	Generic name Brigatinib	Lorlatinib	Entrectinib	Osimertinib mesylate	Dacomitinib	Afatinib	Vemurafenib	Dabrafenib

											ed)
New primary malignancies Hemorrhage QT prolongation Gastrointestinal toxicity Uveitis, rash	Severe cutaneous reactions New cutaneous malignancy Hemorrhage, febrile reaction Cardiomyopathy, QT prolongation ILD/pneumonitis Ocular toxicity Venous thromboembolism	New primary malignancies Hemorrhage Cardiomyopathy Photosensitivity, serous retinopathy, retinal vein occlusion Hepatotoxicity Severe dermatologic reaction Rhabdomyolysis	Cardiomyopathy Venous thromboembolism Serous retinopathy, retinal vein occlusion, uveitis ILD/pneumonitis Hepatotoxicity Rhabdomyolysis Hemorrhage	Serious and potentially fatal hepatotoxicity	Neurotoxicity Hepatotoxicity	Neutropenia Gastrointestinal toxicity	QT prolongation Hepatobiliary toxicity Neutropenia	Diarrhea, Hepatotoxicity Neutropenia Venous thromboembolism	n Myelosuppression Secondary MDS/AML Pneumonitis	Secondary MDS/AML	(continu
Melanoma with BRAF V600E or V600K mutation	Melanoma with BRAF V600E or V600K mutation	Melanoma with BRAF V600E or V600K mutation	Melanoma with BRAF V600E or V600K mutation			HR-positive, HER2- negative breast cancer	HR-positive, HER2- negative breast cancer	HR-positive, HER2- negative breast cancer	BRCA-mutated ovarian and breast cancer		
Melanoma	Melanoma	Melanoma	Melanoma	Tenosynovial giant cell tumor	Solid tumors with NTRK gene fusion	Breast cancer	Breast cancer	Breast cancer	Breast cancer Ovarian cancer	Ovarian, fallopian tube, or primary peritoneal cancer	
RAF kinase inhibitor	MEK1/2 inhibitor	MEK1 inhibitor	MEK1/2 inhibitor	KIT, CSF1R, FLT3 inhibitor	TRK inhibitor	CDK inhibitor	CDK inhibitor	CDK inhibitor	PARP inhibitor	PARP inhibitor	
06/27/18	05/29/13	11/10/15	06/27/18	08/02/19	11/26/18	02/03/15	03/13/17	09/28/17	12/19/14	12/19/16	
BRAFTOVI	MEKINIST	COTELLIC	MEKTOVI	TULARIO	VITRAKVI	IBRANCE	KISQALI	VERZENIO	LYNPARZA	RUBRACA	
Encoraĉenib	Trametinib	Cobimetinib	Binimetinib	Pexidartinib	Larotrectinib sulfate	Palbociclib	Ribociclib	Abemaciclib	Olaparib	Rucaparib camsylate	

(continued)
й
ß
U
9
Ч

	Adverse events	Secondary MDS/AML Myelosuppression Hypertensive crisis	Secondary MDS/AML Myelosuppression	Severe hypersensitivity Severe cutaneous reactions Hyperglycemia, DKA Pneumonitis Severe diarrhea	Severe myelosuppression Gastrointestinal toxicity	Ischemic cardiovascular events, hypertension Fractures and falls Posterior reversible encephalopathy, seizure	Ischemic cardiovascular events Fractures and falls Seizure	Fatigue, pain in extremity, rash	Embryo-fetal death, severe birth defects Muscles spasms Gastrointestinal toxicity	Embryo-fetal death, severe birth defects CPK elevation		Fatal neutropenic sepsis Severe diarrhea ILLD Severe hvnersensitivity	Neutropenic sepsis Rhabdomyolysis Hepatotoxicity Cardiomyopathy Extravasation with tissue necrosis
	Comments		BRCA-mutated, HER2-negative breast cancer	HR-positive, HER2- negative, PIK3CA- mutated breast cancer		CRPC	mCSPC, nmCRPC	nmCRPC					
	Indication	Ovarian, fallopian tube, or primary peritoneal cancer	Breast cancer	Breast cancer	Esophageal cancer Gastric cancer Colorectal cancer	Prostate cancer	Prostate cancer	Prostate cancer	Basal cell carcinoma	Basal cell carcinoma		Pancreatic cancer	Soft tissue sarcoma
	Mechanism of action	PARP inhibitor	PARP inhibitor	Phosphoinositide 3-kinase (Pl3K) inhibitor	TFT and TPI inhibitor	Androgen receptor antagonist	Androgen receptor antagonist	Androgen receptor antagonist	SMO antagonist	SMO antagonist		Liposomal formulation of irinotecan	DNA binding and cross-linking
FDA approval	date	03/27/17	10/16/18	05/24/19	09/22/15	08/31/12	02/14/18	07/30/19	01/30/12	07/24/15		10/22/15	10/23/15
	Brand name	ZEJULA	TALZENNA	PIQRAY	LONSURF	XTANDI	ERLEADA	NUBEQA	ERIVEDGE	ODOMZO		ONIVYDE	YONDELIS
	Generic name	Niraparib tosylate monohydrate	Talazoparib tosylate	Alpelisib	Trifluridine and tipiracil hydrochloride	Enzalutamide	Apalutamide	Darolutamide	Vismodegib	Sonidegib	Cytotoxic agents	Irinotecan hydrochloride liposome	Trabectedin

<i>noclonal antiboc</i> tuzumab	ties against cell si PERJETA	urface antigens 06/08/12	HER2 recombinant	Breast cancer	HER2-positive breast	Cardiotoxicity
			humanized IgG1 mAb		cancer	Infusion reaction Hypersensitivity Gastrointestinal toxicity
rastuzumab sine	KADCYLA	02/22/13	HER2 recombinant humanized IgG1 mAb	Breast cancer	HER2-positive breast cancer	Cardiotoxicity Hepatotoxicity Gastrointestinal toxicity Infusion reaction Hypersensitivity
cirumab	CYRAMZA	12/12/14	VEGFR2 recombinant IgG1 mAb	NSCLC Gastric cancer Colorectal cancer Liver cancer		Infusion-related reaction Hypertension Hemorrhage Arterial thrombotic events Reversible posterior leukoencephalopathy Wound healing complications
uximab	UNITUXIN	03/10/15	GD2 mouse/human IgG1 mAb	Neuroblastoma		Infusion reactions, capillary leak syndrome, hypotension Neuropathic pain Ocular neuropathy Myelosuppression Electrolyte abnormalities Hemolytic uremic syndrome
imumab	PORTRAZZA	11/24/15	EGFR human lgG1 mAb	NSCLC		Cardiopulmonary arrest and/or sudden death Hypomagnesemia Venous and arterial thromboembolic events Dermatologic toxicities Infusion reactions
umab dy-drug conjı	LARTRUVO ugates	10/19/16	PDGFR human IgG1 mAb	Soft tissue sarcoma		Infusion reactions
rastuzumab ecan-nxki	ENHERTU	12/20/2019	Humanized anti-HER2 IgG1 mAb covalently linked to a topoisomerase inhibitor	Breast cancer		ILD, pneumonitis Neutropenia LV dysfunction
umab n-ejfv	PADCEV	12/18/2019	Human IgG1 mAb directed against nectin-4, microtubule inhibitor conjugate	Refractory urothelial cancer		Hyperglycemia Peripheral neuropathy Ocular and skin disorders Infusion site extravasation
tbinant fusion libercept	protein ZALTRAP	09/03/12	VEGFR 1 and 2 binding attached to human IgG1	Colorectal cancer		Hemorrhage, GI perforations, fistulas Wound complications Hypertension Thrombotic events Diarrhea Reversible posterior leukoencephalopathy

721

-
 4)
_
_
_
_
_
_
_
_
\sim
<u> </u>
· · · ·
× -
\sim
-
•
_

_
_
-
_

								nsmission and	ptor, <i>HER2</i> humai aplastic lymphoma
Adverse events		Adrenocortical insufficiency Hepatotoxicity Mineralocorticoid excess Cardiovascular disease		Risk from radiation exposure Myelosuppression Secondary malignancies		Risk from radiation exposure Myelosuppression Secondary MDS/leukemia Renal and hepatic toxicity Neuroendocrine hormonal crisis		Accidental exposure may lead to tran herpetic infection Herpetic infection Injection site complications Immune-mediated event Plasmacytoma at injection site	a, FGFR fibroblast growth factor rece TK receptor tyrosine kinase, ALK and
Comments		mCRPC		CRPC		GEP-NETs			The patocellular carcinom owth factor receptor $2, h$
Indication		Prostate cancer		Prostate cancer		Gastroenteropancreatic neuroendocrine turnors		Melanoma	small cell lung cancer, HCC FR2 vascular endothelial gr
Mechanism of action		CYP17 inhibitor		Alpha particle-emitting isotope radium-223		Radiolabeled somatostatin analog		Oncolytic HSV-1	small cell lung cancer, SCLC growth factor recentor. VEC
FDA approval date		04/28/11		05/15/13		01/26/18		10/27/15	m, <i>NSCLC</i> non- <i>GFR</i> epidermal
Brand name		ZYTIGA		XOFIGO		LUTATHERA		IMLYGIC	ag Administratic or receptor $2, E$
Generic name	Antiandrogen	Abiraterone acetate	Radioactive isotope	Radium 223 dichloride	Radioconjugates	Lutetium 177 dotatate	Oncolytic viruses	Talimogene laherparepvec	FDA US Food and Dri enidermal growth fact

kinase, ROSI C-ros oncogene 1, TRK tropomyosin receptor kinases, MEK mitogen-activated protein kinase, CSFIR colony-stimulating factor-1 receptor, FLT3 FMS-like tyrosine kinase 3, CDK cyclin-dependent kinase, PARP poly(ADP-ribose) polymerase, CPK creatine phosphokinase, ILD interstitial lung disease, CNS central nervous system, CHF congestive heart failure, MDS myelo-dysplastic syndrome, mCSPC metastatic castration-sensitive prostate cancer, mmCRPC nonmetastatic castration-resistant prostate cancer





immune-related adverse events, LFTs liver function tests, PE pulmonary embolism, PRES posterior reversible encephalopathy syndrome, SCT stem cell transplantation, SOS sinusoidal occlusion syndrome. *"Torsade de pointes* may be precipitated by QT-prolonging medications or hypomagnesemia. ^sWhen considering the H's in pulseless arrest, assess for hyperkalemia due to tumor lysis syndrome

"Familiar Scenes with New Actors"

Differentiation Syndrome

Differentiation syndrome (DS), also called the retinoic acid syndrome, was first recognized as a complication of treatment with all-trans retinoic acid and/or arsenic trioxide for acute promyelocytic leukemia. This life-threatening emergency is characterized by fever, hemodynamic instability, pulmonary infiltrates or effusions, edema, and renal failure within the first month of induction therapy in 25% (range: 2-48%) of acute promyelocytic leukemia cases [1-3]. In addition, DS can occur in a different clinical context: patients with relapsed or refractory acute myeloid leukemia who are receiving isocitrate dehydrogenase inhibitors (IDH inhibitors) [1]. Ivosidenib (IDH1 inhibitor) causes DS in 10.6% of patients [1] and enasidenib (IDH2 inhibitor) in 11.7% [1, 4]. These are lower rates than in acute promyelocytic leukemia. Classic DS has a bimodal distribution with peak incidence in the first and third weeks after initiation of induction therapy for acute promyelocytic leukemia, but IDH inhibitor-induced DS has a median onset time of 48 days (range 10-340 days) for enasidenib and 29 days (range 5-59 days) for ivosidenib [1, 5]. Thus, IDH inhibitor-induced DS may occur 1 week to 5 months after treatment initiation. Furthermore, IDH inhibitor-induced DS can recur after therapy interruption or dose escalation [4].

DS involves migration and infiltration of differentiated white blood cells into various organs, leading to endothelial activation and release of cytokines and vascular factors [1]. DS is a challenging clinical diagnosis because there are no specific laboratory tests or imaging modalities. Classic signs are dyspnea, unexplained fever, >5-kg weight gain (from fluid retention), unexplained hypotension, acute renal failure, pulmonary infiltrates, and pleural and/or pericardial effusion on imaging [2, 3]. DS grading is based on the number of classic signs and symptoms (Table 53.3). Because DS can mimic decompensated heart failure, pneumonia, and

 Table 53.3
 Differentiation syndrome signs, symptoms, and grading
 [2, 4]

Signs and symptoms	Grading	
Bone pain	\leq 2 signs or symptoms	Mild
Dyspnea		
Fever		
Headache		
Hepatotoxicity	3 signs or symptoms	Moderate
Hypotension		
Leukocytosis		
Peripheral edema		
Pleuropericardial effusion	\geq 4 signs or symptoms	Severe
Pulmonary infiltrates		
Renal failure		
Weight gain >5 kg		

sepsis, emergency physicians should be aware of the clinical settings that raise suspicion of DS in order to promptly diagnose and manage DS. Early intervention and treatment with corticosteroids can reduce mortality from 30% to <1% [2, 3].

Elevated white blood cell count >5 \times 10⁹/L and serum creatinine level are associated with an increased risk of severe DS and mortality [2]. In the right clinical settings, DS should be treated as soon as suspected because delayed workup and management increases morbidity [1, 6]. The role of the emergency physician is critical for early diagnosis and timely management (including a timely hematology/oncologic consultation). The mainstay treatment is high-dose glucocorticoids (dexamethasone 10 mg/day) until symptom resolution [1, 3, 4]. The addition of broad-spectrum antibiotics is appropriate when sepsis is suspected because of the overlap with the signs and symptoms of DS. Uric acid must be checked, and hyperuricemia management is important [1, 2, 4]. For patients with renal failure, hemodialysis may be required, and for those without renal failure, fluid overload can be treated with diuretics [4]. In severe cases, oxygen supplementation, ventilatory and hemodynamic support, and admission to the intensive care unit may be required.

Infusion-Related Reactions

An infusion-related reaction is an adverse reaction to the infusion of a pharmacologic or biologic substance as defined in the National Cancer Institute's Common Terminology Criteria for Adverse Events (CTCAE, version 5.0) [7], and it occurs upon infusion or within a few hours after infusion of a drug (commonly a monoclonal antibody (mAb) and a biological) [8]. CTCAE provides a severity grading scale [9]. Many chemotherapy agents administered intravenously are associated with infusion-related reactions. Infusion reactions can be allergic (IgE-mediated) or anaphylactoid in nature (i.e., not truly allergic and non-IgE-mediated). Both types of reactions involve the immune system and can have a similar clinical presentation: rash, urticaria, pruritus, bronchospasm, nausea, vomiting, abdominal pain, dizziness, headache, throat irritation, oropharyngeal pain, dyspnea, pharyngeal or laryngeal edema, flushing, fever, hypotension, and tachycardia [10]. This combination of signs and symptoms is related to cytokine release within the first few hours after administration. Infusion-related reactions can manifest hours after the patient has left the outpatient infusion facility, requiring presentation to an ED for acute management [11]. Severe reactions require prompt evaluation and management to avoid severe morbidity and mortality [10]. Unlike hypersensitivity reactions, symptoms of these reactions seem to become less intense with subsequent doses.

For the new biologic drugs (see Tables 53.1 and 53.2), infusion-related reactions are usually mild to moderate

(grade \leq 2). The incidence rates of infusion-related reactions appear to vary among these agents. Blinatumomab is a bispecific T-cell engaging molecule that is administered as a continuous infusion for several weeks for patients with acute lymphoblastic leukemia. Infusion-related reactions occur in about half of these patients, but only 0.5% are serious reactions [11, 12]. Ofatumumab and obinutuzumab are associated with around 60% incidence [11, 13]; daratumumab and elotuzumab, 40–50% [11, 14]; mogamulizumab, 34% [15]; brentuximab, 11–15%; inotuzumab ozogamicin, 10% [16]; and pembrolizumab and nivolumab, 1–5% [11].

Management depends on the severity. In the emergency setting, infusion of the triggering agent should be stopped if not already completed. Treatment includes corticosteroids, antihistamines (both H1 and H2 receptor antagonists), and epinephrine (subcutaneous or intramuscular anaphylaxis doses). Stabilize the respiratory status and support hemodynamics according to resuscitation guidelines [11, 17].

QT Prolongation

Acquired QT prolongation on an electrocardiogram is a potentially serious adverse effect of medication. A long OT interval increases the risk for polymorphic ventricular tachycardia (torsade de pointes) and sudden death. The FDA recommends to correct the QT interval using the Fridericia formula (QTc), and cautions against a OTc >450 ms, and especially if >500 ms [18]. QT prolongation is most commonly associated with electrolyte derangements (especially hypomagnesemia and hypokalemia in cancer patients), heart disease, certain antiarrhythmic agents, antiemetic drugs (e.g., ondansetron), antibiotics (e.g., ciprofloxacin), antipsychotics, and antidepressants [19]. It is also associated with multiple drugs used for treatment of malignancies and cancer-associated symptoms and complications [20]. Several new drugs (ivosidenib, nilotinib, belinostat, carfilzomib, and inotuzumab ozogamicin) for hematologic malignancies have QT-prolonging side effects (see Table 53.1) and quite a number (vandetanib, lenvatinib, crizotinib, ceritinib, entrectinib, osimertinib, vemurafenib, dabrafenib, encorafenib, trametinib, and ribociclib) for solid tumors (see Table 53.2).

In general, a thorough medical history (including current use of QT-prolonging medications), physical examination, and electrocardiogram should be obtained for ED patients taking these novel agents who have significant electrolyte abnormalities or symptoms of palpitations, dizziness, or syncope. In many electronic medical record systems or medication prescribing systems, concurrent use of QT-prolonging drugs will trigger warning messages. Caution is advised when QT-prolonging drugs are used together. Vigilance should be maintained to avoid co-administration of QT-prolonging drugs in the ED. Management of significant QT prolongation starts with discontinuation of QT-prolonging agents, correction of electrolyte abnormalities, assessment of patient stability, and cardiologist consultation when deemed necessary. In the setting of cardiac arrest, resuscitation and treatment of arrhythmias are guided by the advanced cardiac life support (ACLS) protocols [9], keeping in mind that intravenous magnesium sulfate would be indicated for *torsade de pointes*.

Sinusoidal Occlusion Syndrome

Sinusoidal occlusion syndrome (SOS), also called hepatic veno-occlusive disease (HVOD), is caused by a prothrombotic hypofibrinolytic state causing obliterative terminal venulitis in the liver. SOS is a serious emergency that happens in 8-14% of patients during the first month after hematopoietic stem cell transplantation. Antibody-drug conjugates, gemtuzumab ozogamicin and inotuzumab ozogamicin (see Table 53.1) are humanized anti-CD33 and anti-CD22 mAb, respectively, conjugated to a potent DNA-binding cytotoxic antibiotic ozogamicin (a derivative of calicheamicin) [21]. With these new drugs, SOS occurs in about 9% of patients with acute myeloid leukemia who receive gemtuzumab ozogamicin [22–24], around 1% in patients with relapsed or refractory B-cell non-Hodgkin lymphoma who receive inotuzumab ozogamicin, and as high as 13% in patients with relapsed or refractory acute lymphoblastic leukemia who receive inotuzumab ozogamicin [25, 26]. The triad of jaundice, hepatomegaly, and ascites should raise suspicion of SOS in these clinical settings.

The clinical presentation typically includes hyperbilirubinemia, elevated liver enzyme level, weight gain, ascites, and tender hepatomegaly. The differential diagnosis includes graft-versus-host disease, Budd-Chiari syndrome, congestive heart failure, medication toxicity, viral hepatitis, and sepsis. Grading is based on clinical findings, such as bilirubin level, liver enzyme levels, weight gain due to fluid retention, increased serum creatinine, and rate of progression (Table 53.4) [27]. With multi-organ failure, 80% of patients may die.

Table 53.4 Clinical grading of sinusoidal obstruction syndrome [27]

Grading criteria	Mild	Moderate	Severe
Total bilirubin, mg/ dL	<5	5-8	>8
Liver enzymes (AST, ALT)	<3× ULN	3–8× ULN	>8× ULN
Serum creatinine	Normal	<2× ULN	>2× ULN
Weight above baseline	<2%	2–5%	>5%
Rate of clinical progression	Slow (6–7 days)	Moderate (4–5 days)	Fast (2–3 days)

AST aspartate transferase, ALT alanine transferase, ULN upper limit of normal

Treatment of SOS is mainly supportive. The main challenge for emergency physicians is fluid management, because intravascular volume needs to be optimized with crystalloids or colloid (e.g., albumin) solutions without overload to avoid hepatorenal syndrome. Colloid solutions may be indicated in the presence of hypoalbuminemia. Adequate oxygenation and maintenance of oxygen-carrying capacity by transfusion would help to minimize hepatic ischemic injury. Nephrotoxic and hepatotoxic drugs should be avoided. In moderate to severe cases, high-dose glucocorticoids (methylprednisolone 1 mg/kg/day intravenously) are indicated [28].

Emergency physicians must be aware of the potential for SOS after treatment with these antibody-drug conjugates or in patients who have recently undergone hematopoietic stem cell transplantation. Early suspicion of SOS is vital given the triad of jaundice, hepatomegaly, and ascites in the above clinical settings. The primary role of the emergency physician is to recognize and diagnose this syndrome and initiate glucocorticoid therapy along with a hematologist/oncologist consultation.

Tumor Lysis Syndrome from Venetoclax-Based Therapy

Tumor lysis syndrome is a well-known oncologic emergency, especially when the tumor/malignancy burden is high. Venetoclax, a BCL2 inhibitor used as a single agent or in combination with other targeted therapies (see Table 53.1), has shown an increase risk of tumor lysis syndrome [29].

Novel Adverse Effects of New Immunotherapy for Cancer

Immune-Related Adverse Events (irAEs)

Immunotherapy has emerged as an effective cancer treatment by enhancing the ability of immune cells to kill cancer cells [30]. Malignant cells can evade host immune surveillance by usurping immune checkpoint pathways. Immune checkpoints regulate the immune response to prevent hyperinflammation or autoimmunity [31–35]. mAbs that bind and block cytotoxic T lymphocyte-associated protein 4 (CTLA4), programmed cell death protein-1 (PD-1), and programmed cell death ligand 1 (PD-L1) are approved to treat many different cancers by "releasing the brakes" on T-cell activation. Immune checkpoint inhibitors are now used for many solid tumors and hematologic malignancies (see Tables 53.1 and 53.2).

Stimulating the immune system to fight cancer is a double-edged sword; it can cause autoimmunity, leading to

many irAEs. The estimated incidence of irAEs of any severity/grade was about 30% for anti-PD-L1 and anti-PD-1 therapies, and 75% for anti-CTLA4 therapy [36]. Most irAEs affect the integumental, gastrointestinal, endocrine, pulmonary, and musculoskeletal systems. Serious renal, neurologic, and cardiac irAEs may also occur. Prompt investigation of symptoms is vital for timely diagnosis and management.

Emergency physicians are at the front line to deal with acute presentation of irAEs. Existing guidelines from various professional organizations (National Comprehensive Cancer Network [NCCN] [37], American Society of Clinical Oncologic [ASCO] [38], Society for Immunotherapy of Cancer [SITC] [39], and European Society for Medical Oncologic [ESMO] [40]) can provide a framework for management of irAEs.

In the ED setting, the medical history about ICI therapy and onset of symptoms may cue the investigation of potential irAEs, since the incidence of irAEs usually peaks around the fourth dose (12–16 weeks after initiation) [41]. However, irAEs can occur any time during treatment (from after a single dose [42] to a prolonged period [40] and even after treatment discontinuation [43–46]). Grading of the severity of irAEs is based on CTCAE, version 5.0 [7]. Table 53.5 compiles grading and grade-based treatment considerations in the ED for common irAEs.

Immunosuppression with a glucocorticoid is the primary treatment for clinically significant irAEs. Escalation to other immunosuppressants (e.g., infliximab for colitis [47] or mycophenolate mofetil for hepatitis [48]) may be required in steroid-refractory cases or when significant steroid-induced side effects occur. Emergency physicians will primarily be dealing with the decision to initiate first-line therapy with glucocorticoids, in collaboration with the primary oncologists [37–39]. The occurrence of irAEs and subsequent use of immunosuppressive therapy do not seem to affect cancer response rates or antitumor therapeutic activity [49, 50].

Cardiac irAEs Cardiac dysfunction induced by ICI results from damage by lymphocytic infiltration [51–55]; immunemediated myocarditis may present in EDs as acute lifethreatening events. The onset may be rapid and can lead to cardiac failure or dysrhythmias (including ventricular tachycardia or complete atrioventricular heart blocks) that are potentially fatal [51, 56, 57] (see Fig. 53.1). The incidence of cardiac dysfunction is not clear but is perhaps $\leq 1\%$ [51, 52]. The acute management of cardiac irAEs basically follows the ACLS algorithms along with consideration for initiation of glucocorticoid therapy emergently and other considerations mentioned in Fig. 53.1. These patients will often need to be admitted for cardiac monitoring, and intensive care needs to be considered.

Endocrine irAEs Endocrine toxicities of ICIs manifest hormone deficiency or excess. The spectrum of clinical ill-

Table 53.5	Management of co	ommon irAEs bas	sed on grades in t	the emergency	department* [37]	I
			8			

	Cristi	1	2	3	4 (1:fe (here)		
	Grade	(mild) Mild or no	(moderate)	(severe) Medically significant:	(life-threatening)		
	guidance for grading	symptoms	ADL	limiting self-care; need hospitalization	to prevent death		
General management ^a							
	Systemic corticosteroid		Prednisone equivalent (0.5–1 mg/kg/day)	Prednisone equivalent (1–2 mg/kg/day)	Prednisone equivalent (1–2 mg/kg/day)		
	ED disposition	Discharge	Consider hospitalization	Hospitalize	Hospitalize; ICU if indicated		
Specific consider	rations						
Maculopapular	Grading	<10% BSA	10–30% BSA	>30%BSA			
rasn	Management	Antihistamine Emollient Topical corticosteroid (moderate potency)	Antihistamine; emollient; topical corticosteroid (high potency)	orticosteroid (high potency)			
Pruritus	Grading	Mild and/or localized	Widespread and intermittent; skin changes from scratching	Widespread and constant; limiting self-care or sleep			
	Management	Topical corticosteroid (high potency)	Dermatology consult; antihistamine; topical corticosteroid (high potency); systemic corticosteroid not needed	Urgent dermatology consult; prednisone (0.5–1 mg/kg/day); GABA agonist (gabapentin/ pregabalin); measure serum IgE and histamine; if high, treat with omalizumab and antihistamine, respectively			
Blistering dermatological	Grading			Skin sloughing <10% BSA in SJS	Skin sloughing 10–30% BSA in SJS or ≥30% BSA in TEN		
disorders (bullae) ^b	Management	Urgent dermatology consult; skin biopsy; topical corticosteroid (high potency)	Urgent dermatology consult; skin biopsy	Urgent dermatology & ophthalmology consults; skin biopsy	Urgent dermatology & ophthalmology consults; skin biopsy		
Diarrhea/colitis	Grading	<4 BM/day above baseline & no symptoms of colitis	4–6 BM/day above baseline & colitis symptoms; no interference of ADL	>6 BM/day above baseline & colitis symptoms interference of ADL	>6 BM/day above baseline & colitis symptoms; interference of ADL		
	Management	Hydration Antidiarrheals	Gastroenterology consult; CT scan of the abdomen/pelvis; r/o infection; inpatient supportive care; methylprednisolone (1 mg/kg/day IV)	Gastroenterology consult; CT scan of the abdomen/ pelvis; r/o infection; inpatient supportive care; methylprednisolone (2 mg/ kg/day IV)	Gastroenterology consult; CT scan of the abdomen/ pelvis; r/o infection; inpatient supportive care; methylprednisolone (2 mg/ kg/day IV)		
Transaminitis	Grading	<3× ULN	35× ULN	>5-20× ULN	>20× ULN		
(hepatic)	Management	Consider gastroenterology consult; r/o other causes (e.g., viral infection); discontinue hepatotoxic drugs	Consider gastroenterology consult; r/o other causes; discontinue hepatotoxic drugs; if bilirubin >1.5× ULN, hepatology consult, inpatient care, and prednisone equivalent (2 mg/kg/day)	Hepatology consult; r/o other causes; discontinue hepatotoxic drugs; prednisone equivalent (1–2 mg/kg/day); if bilirubin >1.5× ULN, prednisone equivalent (2 mg/kg/day)	Hepatology consult; r/o other causes; discontinue hepatotoxic drugs; methylprednisolone (2 mg/ kg/day IV)		

(continued)

Table 53.5 (continued)

		1	2	3	4
	Grade	(mild)	(moderate)	(severe)	(life-threatening)
Pneumonitis	Grading	No symptoms; confined to one lobe or <25% lung tissue	New and worsening symptoms	Involving all lobes or >50% lung tissue; limited self-care ADL	Life-threatening respiratory compromise
	Management	Pulse oximetry; chest imaging	Chest imaging; infection workup; consider bronchoscopy; supplemental oxygen; prednisone equivalent 1–2 mg/kg/day; empiric antibiotics	Chest imaging; infection workup; consider bronchoscopy; supplemental oxygen; prednisone equivalent 1–2 mg/kg/day; empiric antibiotics; pulmonary and infectious disease consults	Chest imaging; infection workup; consider bronchoscopy; supplemental oxygen/mechanical ventilation; fluid management; prednisone equivalent 1–2 mg/kg/day; empiric antibiotics; pulmonary and infectious disease consults

ADL activity of daily living, BSA body surface area, ICU intensive care unit, SJS Steven-Johnson syndrome, TEN toxic epidermal necrolysis, ULN upper limit of normal

^aGenerally applicable unless otherwise stated in special considerations below

^bSteven-Johnson syndrome and toxic epidermal necrolysis are considered grade 3–4 bullous dermatitis

*The specific considerations shown in this table only cover some common irAEs. Please see the NCCN guidelines for more details

ness ranges from no symptoms to life-threatening endocrine emergencies [58–63] (e.g., thyroid storm [64], myxedema coma [65], diabetes ketoacidosis [66]) (see Fig. 53.1). ED visits for endocrine irAEs are often related to the thyroid, adrenal, and pituitary glands [67].

Hypophysitis often presents with nonspecific symptoms (headache, fatigue, and general weakness) that are often misattributed to malignancy, but failure to recognize and diagnose hypophysitis can lead to adrenal crisis [68-70]. Anorexia, nausea, weight loss, altered mental status, heat intolerance, and arthralgia are less common symptoms (10.5–21.1%) [68, 70, 71]. Visual defects, typically bitemporal hemianopsia, are rare because pituitary enlargement is mild [72]. Hyponatremia is a major electrolyte disorder associated with hypophysitis (47–56% of patients) [70, 71, 73]. Morbidity is predominantly related to central adrenal insufficiency [74], and can be fatal if undiagnosed and untreated. Initial treatment includes high-dose glucocorticoids (1 mg/kg methylprednisolone or equivalent) along with management of hyponatremia and hypotension. Without significant hyponatremia, intense headaches, or optic chiasm compression, physiologic replacement of glucocorticoids may be considered [75].

Primary adrenal insufficiency associated with ICIs is uncommon but may be underreported due to concurrent treatment with glucocorticoids for various reasons (including oncologic indications) or coexistence of central adrenal insufficiency [76]. Symptoms of adrenal crisis include hypotension, dehydration, and electrolyte abnormalities (especially hyponatremia), which require immediate intervention [74]. These symptoms can improve rapidly after starting corticosteroids [70]. Hydrocortisone and fludrocortisone are the mainstay treatments [77].

Most cases of primary thyroid irAE are related to thyroiditis [78]. Thyroiditis can manifest as subclinical or overt hyper- or hypothyroidism with a typical clinical course of transient hyperthyroidism (due to thyroid hormones released by gland destruction) followed by prolonged or permanent hypothyroidism [78, 79]. Graves' disease is an autoimmune hyperthyroidism in which autoantibodies stimulate the thyroid-stimulating hormone receptor. Thyrotoxic cancer patient may present to the ED for arrhythmia (e.g., atrial fibrillation, atrial flutter, and supraventricular tachycardia), but these patients with arrhythmia will also need to be evaluated to rule out myocarditis and pericarditis as concurrent irAEs contributing to arrhythmia. Thyrotoxicosis in thyroiditis is usually mild and manageable with β-blockers. However, in patients with Graves' disease, life-threatening thyroid storms can occur. Serum thyroid-stimulating antibodies and radionuclide thyroid scans can distinguish Graves' disease from thyroiditis. Graves' disease can be managed with β -blockers, antithyroid agents, and glucocorticoids [80] and, if needed, additional agents that block thyroid hormone release (e.g., potassium iodide, iopanoic acid, etc.) [81].

Immune thyroiditis may destroy enough thyroid tissue to cause primary hypothyroidism. Hypothyroid symptoms are often nonspecific [82]. Typical complaints include fatigue, lethargy, constipation, and cold intolerance. Unrecognized severe hypothyroidism may progress to myxedema coma. If a cancer patient with a history of immune checkpoint inhibitor therapy is in the ED with confusion, lethargy, bradycardia, hyponatremia, hypothermia, or constipation, severe hypothyroidism should be included in the differential diagnosis. Primary hypothyroidism is primarily managed with thyroid hormone replacement (e.g., levothyroxine) and supportive care [83].

New-onset type 1 diabetes mellitus may present with diabetic ketoacidosis in patients receiving ICI therapy [84–86], and can occur even after one dose [42]. Although rare, diabetic ketoacidosis is potentially life-threatening and requires early recognition to initiate timely management including intravenous insulin infusion. Once stabilized, the patient requires long-term, if not lifelong, insulin therapy [87].

Gastrointestinal irAEs Patients treated with combination ICI therapy had incidence rates of 13.6%, 9.4%, and 9.2% for all-grade colitis, severe colitis, and severe diarrhea, respectively [88], which are higher rates than in monotherapy. Symptoms typically start 6-8 weeks after beginning ICI therapy [89, 90]. Corticosteroids are the primary treatment for gastrointestinal toxicities, with escalation to infliximab (mAb against tumor necrosis factor-alpha) for severe steroidrefractory colitis [37, 91-93]. Vedolizumab (mAb against $\alpha 4\beta 7$ integrin) can be effective in steroid-refractory and infliximab-resistant enterocolitis [94, 95]. Immune-related hepatotoxicity is usually mild, presenting with no or a few nonspecific symptoms, but can be severe or even fatal in some cases [96]. Median time to hepatitis is typically 5–6 weeks after starting therapy but can occur months later. The differential diagnosis should be broad, and infectious hepatitis must be ruled out, especially when the patient also has fever. Autoimmune and drug-induced hepatitis can have similar presentations and may require further imaging studies and/or biopsy.

Hemophagocytic lymphohistiocytosis (HLH) HLH is also called macrophage activation syndrome (MAS) and is a life-threatening syndrome of hyperinflammation and progressive immune-mediated organ damage. ICI-induced HLH is rare. HLH is diagnosed by having >5 of the following criteria: (1) fever; (2) splenomegaly; (3) cytopenias (at least two lineages in complete blood counts); (4) hypertriglyceridemia or hypofibrinogenemia; (5) hemophagocytosis; (6) low natural killer cell function; (7) high serum ferritin; and (8) high-soluble IL-2 [97, 98]. Some of the diagnostic criteria will not be able to be assessed in the emergency center, but HLH should be considered in the differential diagnosis for cancer patients treated with ICIs presenting with fever, cytopenias, and signs of hyperinflammation [99] (see Fig. 53.1). Optimal treatment for HLH caused by immunotherapy is not known. In two case reports of HLH, one patient was treated with etoposide and dexamethasone [100], and the other was treated with prednisone and mycophenolate mofetil [101]. Morbidity and mortality due to HLH are partially attributed to delayed diagnosis, hence the vitally important role of the emergency physician in achieving early diagnosis and intervention with high-dose glucocorticoids.

Integumental irAEs IrAE involving the subcutaneous tissue is relatively rare, manifesting as varying forms of panniculitis [102–104]. In contrast, cutaneous irAEs are the

most common irAEs, occurring in almost half of the patients receiving ICI therapy (Table 53.2). The majority are low grade, presenting within the first two cycles of therapy [105–109] with itching, erythema, rash, vesicles, or blisters [105, 108, 110, 111]. High-grade cutaneous irAEs occur in about 1–3% of patients, and require ICI therapy discontinuation and treatment with high-dose corticosteroids; GABA agonists (e.g., pregabalin or gabapentin) can be used for severe pruritus [38, 106, 110, 112]. Serious irAEs such as toxic epidermal necrolysis (TEN), drug rash with eosinophilia and systemic symptoms (DRESS), and Steven-Johnson syndrome (SJS) can be fatal if not recognized and treated with high-dose corticosteroids as inpatient [37, 113–115].

Neurologic irAEs Low-grade and nonspecific neurologic symptoms, such as headache, dizziness, and sensory impairment, occur in 6-12% of patients receiving ICI combination or monotherapy [45, 116]. About 0.4–0.2% of patients treated with nivolumab and pembrolizumab have grade ≥ 3 irAEs [116, 117]; 0.3–0.8%, with ipilimumab [118, 119]; and up to 0.7%, with combination nivolumab and ipilimumab [120]. Overall, severe neurologic irAEs is rare (<1% of patients) [45], and usually happens within 1-7 weeks of therapy [121-123]. Seizures, confusion, ataxia, abnormal behavior, and altered consciousness warrant evaluation for encephalitis which occurs in 0.1–0.2% of patients [116, 124– 127] (see Fig. 53.1). Emergent brain imaging with CT or MRI with and without contrast if available in the ED is indicated. Brain MRI is the imaging modality of choice for evaluation of neurologic irAEs. Non-enhanced MRI shows T2 hyperintensity in a quarter of cases [124, 125]. Dural thickening and meningeal enhancement may also be present [127]. If fever is present together with headache and/or meningeal signs, lumbar puncture with analysis of cerebrospinal fluid (CSF) in the ED is indicated to rule out meningitis. In neurologic irAEs, CSF analyses are consistent with lymphocytic meningitis (with mild to high pleocytosis, negative cultures, elevated protein, and cytopathology negative for malignancy).

Pulmonary irAEs The most frequently reported pulmonary symptoms are dyspnea (see Fig. 53.1) and nonproductive cough, with fever and chest pain reported less commonly [113, 128]. The presence of extrapulmonary irAEs is relatively common and should raise suspicion for coexisting pneumonitis [129] (see Table 53.2). Generally, new persistent cough or shortness of breath in patients receiving immune checkpoint inhibitors should prompt evaluation for pneumonitis. The incidence of pneumonitis is <1% but varied across trials of immune checkpoint inhibitors, perhaps because of underreporting of low-grade irAEs [130]. Nevertheless, pneumonitis is one of the most serious irAEs and can be fatal [74, 128, 131–133]. In patients with acute

pulmonary symptoms, infectious pneumonia is in the differential diagnosis, and empiric antibiotics coverage is indicated. Evaluation of suspected pneumonitis should include a CT scan of the chest and possibly, after admission from the ED, a bronchoalveolar lavage and lung biopsy [134]. Patients with grade ≥ 2 pneumonitis should start glucocorticoids, and be closely monitored [38]. Grade \geq 3 pneumonitis requires inpatient care, evaluation to exclude infections, and probably consultation with pulmonary and infectious disease experts. Treatment with high-dose glucocorticoids (methylprednisolone, 1–2 mg/kg per day) should be initiated after discussion between the emergency physician and the oncologist, aiming to improve symptoms down to grade ≤ 1 . Other immunosuppressant agents (e.g., infliximab, mycophenolate mofetil) or intravenous immunoglobulins are generally not used in the ED but may be used after admission if not improved after 48 hours.

Rheumatologic irAEs Rheumatic manifestations in cancer patients receiving immune checkpoint inhibitors are primarily arthritis, sicca symptoms (dry eyes, dry mouth, and parotid gland enlargement), and a polymyalgia-like syndrome [44, 135]. Among irAEs involving the musculoskeletal system, myositis can be serious. Myositis presents as proximal muscle weakness with an increase in serum muscle enzymes, with or without respiratory complaints [56, 136– 148]. De novo cases have been reported [136, 149]. Necrotizing myositis is an emerging irAE [136, 146, 147, 150-152]. Several cases of concomitant myositis and myasthenia-like manifestations have been documented [136, 145]. Patients often present with "dropped heads," bilateral proximal limb muscle weakness, muscle pain, dyspnea, and, occasionally, fever. Of note, myositis can also be associated with unilateral or bilateral ptosis, ophthalmoparesis, and bulbar weakness. Creatine phosphokinase and aldolase levelsand sometimes cardiac troponins-are elevated. Further diagnostic evaluation may involve acetylcholine receptor antibody and myositis serum antibodies and electromyography, which usually occur after admission from the ED. The time from initiation of immunotherapy to development of myositis is variable, but in most cases, myositis occurs after a few infusions. Myositis can be life-threatening because of CO₂ retention and respiratory failure, and severe rhabdomyolysis may cause renal failure, electrolyte imbalance, and fatal arrhythmia; both requiring prompt diagnosis and management. When respiratory compromise is suspected (as prompted by shallow rapid breathing, dyspnea, low oxygen saturation, etc.), ventilatory status should be evaluated with arterial blood gas measurement, peak flows, and/or bedside pulmonary function tests (negative inspiratory force and vital capacity). If patient has respiratory failure, support with positive pressure ventilation or intubate for mechanical ventilation.

Chimeric Antigen Receptor (CAR)-T-Cell Therapy

Genetically engineered T-cells constitute a new class of therapeutic agents that offer hope for cure of malignancies. CD19-CAR-T-cell therapy was recently approved for the treatment of hematologic malignancies (see Table 53.1). CAR-T-cell therapy causes a new spectrum of acute toxic effects, which differ from those typically organ-specific irAEs associated with ICIs [30, 41, 153]. CAR-T-cell therapy-related toxicities are related to markedly increased circulating cytokines and manifest as two syndromes: cytokine release syndrome (CRS) and immune effector cellassociated neurotoxicity syndrome (ICANS) [12, 154, 155]. Occasionally, CRS can progress into HLH [12, 156, 157]. Glucocorticoids are immunosuppressive and effective in cellular therapy-induced CRS, ICANS, and HLH. Even though ED visits for these adverse events are rare as these patients are typically hospitalized during treatment, this landscape is likely to change as clinical experience with CAR-T-cell therapies increases and the duration of hospitalization will shorten.

Cytokine release syndrome Symptoms of CRS range from mild "flu-like" to life-threatening inflammatory responses (Table 53.6) [158]. Symptoms include high-grade fever, hypotension, hypoxia, fatigue, arthralgia, myalgia, rash, and/ or multi-organ toxicity, which can also be symptoms of infection or sepsis (see Fig. 53.1). Serious CRS is characterized by hypotension and hyperpyrexia that can progress to an uninhibited systemic inflammatory response syndrome with circulatory shock requiring vasopressors, vascular leak, disseminated intravascular coagulation, and multi-organ system failure. Laboratory abnormalities include cytopenias, elevated creatinine, elevated liver enzymes, impaired coagulation, and elevated C-reactive protein level. The primary differential diagnosis is sepsis and septic shock, and empiric antibiotic coverage is recommended.

Respiratory involvement is common with CRS. Mild cases may involve tachypnea and cough but may progress to hypoxemia with bilateral pulmonary infiltrates on chest radiography. Some cases can progress to acute respiratory distress syndrome requiring mechanical ventilation. Moreover, patients with serious CRS often display vascular leak with peripheral and pulmonary edema and may have signs of cardiac dysfunction with reduced left ventricular contractility.

Management of CRS varies by grade [155, 159] (Fig. 53.2). For grade 1, supportive care should keep the patient well-hydrated using intravenous fluids with special attention to fluid balance to avoid pulmonary vascular congestion. For grade 2, hypotension should be treated promptly with intravenous crystalloid fluid. In patients who

Table 53.6	American Sc	ociety for	Blood and	Marrow	Transplantat	tion cytokine	e release syndrome	consensus grading ^a
		2			4	~	2	0 0

	Grade			
Parameter	1	Grade 2	Grade 3	Grade 4
Fever ^b	Present	Present	Present	Present
		Accompanied by		
Hypotension	None	Vasopressors not required And/or ^c	Vasopressor with/without vasopressin	Hypotension requiring multiple vasopressors any except vasopressin
Нурохіа	None	Hypoxia requiring low-flow nasal cannula ^d or blowby	Hypoxia requiring high-flow nasal cannula, face mask, non-rebreather mask, or Venturi mask	Hypoxia requiring positive pressure ^e

Adapted from Lee et al. [158], with permission from Elsevier

^aOrgan toxicities associated with CRS may be graded according to CTCAE v5.0, but they do not influence this CRS grading

^bFever as temperature \geq 38 °C not attributable to other case

°CRS grade determined by the more severe event

^dLow-flow nasal cannula as oxygen ≤6 L/min. High-flow nasal cannula is defined as oxygen >6 L/min

°CPAP, BiPAP, intubation, and mechanical ventilation



Fig. 53.2 Management algorithm for cytokine release syndrome (CRS). LVEF left ventricular ejection fraction, ULN upper limit of normal. (Adapted from Brudno and Kochenderfer [159] with permission from Elsevier)

are refractory to fluid boluses, management should include therapies that block IL-6 signaling (i.e., anti-IL-6 receptor antibody tocilizumab or anti-IL-6 antibody siltuximab) and glucocorticoids. For grade 3 or 4, emergently manage arrhythmias, hemodynamic shock, and respiratory compromise with antiarrhythmics, vasopressors, oxygen, and ventilation support, therapies that block IL-6 signaling and high-dose glucocorticoids. If possible, the emergency team should be aware of all patients treated with CAR-Tcell therapy in the affiliated hospitals to facilitate prompt awareness of their CAR-T-cell therapy status when these patients present to the ED.

Immune effector cell-associated neurotoxicity syndrome Neurotoxicity is the second most serious adverse event following CAR-T-cell therapy [160]. Patients may develop a toxic encephalopathy (immune effector cellassociated encephalopathy [ICE]) with symptoms of confusion/delirium, problems with word retrieval, headache, somnolence, hallucinations, aphasia, hemiparesis, cranial nerve palsies, tremors, and, occasionally, seizures [77, 155– 157, 161–166]. The pathophysiology of neurotoxicity is poorly understood, and is not strictly associated with the timing of CRS [155]. ICANS often starts shortly after CAR-Tcell infusion while the patients are still hospitalized, but some are delayed, and these patients may present to EDs with only neurologic symptoms.

The earliest signs of ICANS are decreased attention span, loss of language coherence, and impaired handwriting (Table 53.7). In severe cases (grade \geq 3), seizures, motor weakness, incontinence, increased intracranial pressure, obtundation, papilledema, and cerebral edema may occur (see Fig. 53.1). The diagnosis of ICANS and evaluation may be guided by the need to obtain the information to use Table 53.7 to obtain the Immune Effector Cell-Associated Encephalopathy (ICE) score for the American Society for Blood and Marrow Transplantation Consensus Grading Scale. Emergent imaging study of the brain using CT or MRI is necessary. Emergent lumbar puncture for cerebrospinal fluid analysis would be indicated in the presence of mental status change and fever (e.g., concurrent CRS) to

examine the differential diagnoses of meningitis and

encephalitis.

Similar to CRS, the management of ICANS is also based on the toxicity grade (Table 53.8). For grade 1, supportive care is provided. Increasing the angle of the head of the bed to $>30^{\circ}$ may minimize the risk of aspiration and improve cerebral venous blood flow. Neurology consultation should be expeditiously requested to facilitate evaluation, including electroencephalogram, evaluation of intracranial pressure, and brain imaging studies. For grade 2, give high doses of glucocorticoids (dexamethasone 40 mg/day or methylprednisolone 1 g/day) until neurologic recovery [167]. For grades 3 and 4, in addition to high-dose glucocorticoids, control seizures and treat status epilepticus. Cerebral edema and increased intracranial pressure may require intensive care [168], and management with mechanical hyperventilation, acetazolamide, and/or mannitol [167]. In patients with concurrent neurotoxicity and CRS, treatment should include tocilizumab (8 mg/kg) and glucocorticoids until resolution of symptoms [167].

Table 53.7 Assessment of immune effect	cell-associated n	neurotoxicity synd	lrome (ICA	ANS))
--	-------------------	--------------------	------------	------	---

Signs and symptoms	Grade 1 (mild)	Grade 2 (moderate)	Grade 3 (severe)	Grade 4 (critical)
ICE ^a : orientation to year, month, city, hospital, 4 points; able to name three objects, 3 points; following simple commands, 1 point; able to write a sentence,1 point; or able to count backwards from 100 in 10s, 1 point	Able to perform 7–9 tasks	Able to perform 3–6 tasks	Able to perform 0–2 tasks	Obtunded and cannot perform tasks
Depressed level of consciousness	Spontaneous awakening	Awakening to voice	Awakening limited to tactile stimulus	Unarousable/requiring repeated or vigorous tactile stimuli for arousal Stupor/coma
Seizures	NA	NA	Clinical seizure focal or generalized resolving rapidly or nonconvulsive seizures on EEG responding to intervention	Life-threatening extended seizure (greater than 5 mins); or repeated clinical or electrical seizures with no return to baseline
Motor findings	NA	NA	NA	Deep focal weakness, e.g., hemiparesis/paraparesis
Increased intracranial pressure/cerebral edema	NA	NA	Focal/local edema on brain imaging	Diffuse cerebral edema on brain imaging; decerebrate or decorticate posturing; or abducens nerve palsy; papilledema; or Cushing triad

Adapted from Lee et al. [158], with permission from Elsevier

NA not applicable, CRS cytokine release syndrome, EEG electroencephalogram

^aFor Immune Effector Cell-Associated Encephalopathy (ICE) score, naming one object counts as one task, so the patient has the opportunity to perform three tasks
Tabl	e 53.8	Management of	f immune effector	cell-associated	l neurotoxicity	syndrome	[167	, 16	58]
------	--------	---------------	-------------------	-----------------	-----------------	----------	------	------	-----

	Grade	Grade 2	Grade 3	Grade 4
Grade	(mild)	(moderate)	(severe)	(critical)
Supportive care ^a	+	+	+	+
Neurology consult	+	+	+	+
Glucocorticoids	NA	Dexamethasone (40 mg/day) or methylprednisolone (1 g/day)	Dexamethasone (40 mg/day) or methylprednisolone (1 g/day)	Dexamethasone (40 mg/day) or methylprednisolone (1 g/day)
Tocilizumab	NA	NA	If concurrent with CRS, tocilizumab (8 mg/kg)	If concurrent with CRS, tocilizumab (8 mg/kg)
Seizure management	NA	NA	+	+
Cerebral edema and ICP management	NA	NA	Acetazolamide and/or mannitol	Mechanical hyperventilation, acetazolamide and/or mannitol
Critical care	NA	NA	+	+

CRS cytokine release syndrome, ICP increased intracranial pressure, NA not applicable

^aSupportive care may include elevating the head of the bed to >30° to minimize the risk of aspiration and improve cerebral venous blood flow

Conclusion

Maintaining a fund of knowledge with updated information about the adverse effects of new cancer therapeutics is important. Knowing the timing, risk, and type of cancer treatment-induced toxicity will arm emergency physicians with the appropriate level of suspicion not to miss the correct diagnosis such that prompt management can improve patient outcomes. Rapid recognition and prompt initiation of appropriate evaluations and treatments are critical actions in the care of these oncologic emergencies.

References

- Abou Dalle I, DiNardo CD. The role of enasidenib in the treatment of mutant IDH2 acute myeloid leukemia. Ther Adv Hematol. 2018;9(7):163–73.
- Montesinos P, Bergua JM, Vellenga E, Rayon C, Parody R, de la Serna J, et al. Differentiation syndrome in patients with acute promyelocytic leukemia treated with all-trans retinoic acid and anthracycline chemotherapy: characteristics, outcome, and prognostic factors. Blood. 2009;113(4):775–83.
- Cardinale L, Asteggiano F, Moretti F, Torre F, Ulisciani S, Fava C, et al. Pathophysiology, clinical features and radiological findings of differentiation syndrome/all-trans-retinoic acid syndrome. World J Radiol. 2014;6(8):583–8.
- Fathi AT, DiNardo CD, Kline I, Kenvin L, Gupta I, Attar EC, et al. Differentiation syndrome associated with enasidenib, a selective inhibitor of mutant isocitrate dehydrogenase 2: analysis of a phase 1/2 study. JAMA Oncol. 2018;4(8):1106–10.
- Stein EM, DiNardo CD, Pollyea DA, Fathi AT, Roboz GJ, Altman JK, et al. Enasidenib in mutant IDH2 relapsed or refractory acute myeloid leukemia. Blood. 2017;130(6):722–31.
- Luesink M, Pennings JL, Wissink WM, Linssen PC, Muus P, Pfundt R, et al. Chemokine induction by all-trans retinoic acid and arsenic trioxide in acute promyelocytic leukemia: triggering the differentiation syndrome. Blood. 2009;114(27):5512–21.
- National Cancer Institute. Division of Cancer Treatment & Diagnosis. CTEP Cancer Therapy Evaluation Program. Common terminology criteria for adverse events (CTCAE). Version 5.0. 2018.

- Bupathi M, Hajjar J, Bean S, Fu S, Hong D, Karp D, et al. Incidence of infusion reactions to anti-neoplastic agents in early phase clinical trials: the MD Anderson Cancer Center experience. Invest New Drug. 2017;35(1):59–67.
- 9. ACLS. Instructor manual eBook. American Heart Association: Dallas; 2019.
- Vogel WH. Infusion reactions: diagnosis, assessment, and management. Clin J Oncol Nurs. 2010;14(2):E10–21.
- Rosello S, Blasco I, Garcia Fabregat L, Cervantes A, Jordan K. Management of infusion reactions to systemic anticancer therapy: ESMO clinical practice guidelines. Ann Oncol. 2017;28(Suppl 4):iv100–iv18.
- Teachey DT, Lacey SF, Shaw PA, Melenhorst JJ, Maude SL, Frey N, et al. Identification of predictive biomarkers for cytokine release syndrome after chimeric antigen receptor T-cell therapy for acute lymphoblastic leukemia. Cancer Discov. 2016;6(6):664–79.
- 13. Ohmachi K, Tobinai K, Kinoshita T, Ishikawa T, Hatake K, Ichikawa S, et al. Efficacy and safety of obinutuzumab in patients with previously untreated follicular lymphoma: a subgroup analysis of patients enrolled in Japan in the randomized phase III GALLIUM trial. Int J Hematol. 2018;108(5):499–509.
- Zhang T, Wang S, Lin T, Xie J, Zhao L, Liang Z, et al. Systematic review and meta-analysis of the efficacy and safety of novel monoclonal antibodies for treatment of relapsed/refractory multiple myeloma. Oncotarget. 2017;8(20):34001–17.
- Kim YH, Bagot M, Pinter-Brown L, Rook AH, Porcu P, Horwitz SM, et al. Mogamulizumab versus vorinostat in previously treated cutaneous T-cell lymphoma (MAVORIC): an international, open-label, randomised, controlled phase 3 trial. Lancet Oncol. 2018;19(9):1192–204.
- Al-Salama ZT. Inotuzumab ozogamicin: a review in relapsed/ refractory b-cell acute lymphoblastic leukaemia. Targeted Oncol. 2018;13(4):525–32.
- 17. U.S. Food & Drug Administration. Center for Drug Evaluation and Research. Center for Biologics Evaluation and Research. E14 clinical evaluation of QT/QTc interval prolongation and proarrhythmic potential for non-antiarrhythmic drugs. Oct 2012. FDA-2004-D-0241. Content Current as of 24 Aug 2018.
- U.S. Food & Drug Administration. Center for Drug Evaluation and Research. Center for Biologics Evaluation and Research. E14 clinical evaluation of QT/QTc interval prolongation and proarrhythmic potential for non-antiarrhythmic drugs questions and Answers (R3). Guidance for Industry. June 2017. FDA-2004-D-0241. Content Current as of 24 Aug 2018.
- Buss VH, Lee K, Naunton M, Peterson GM, Kosari S. Identification of patients at-risk of qt interval prolongation during medication reviews: a missed opportunity? J Clin Med. 2018;7(12):533.

- Becker TK, Yeung S-CJ. Drug-induced QT interval prolongation in cancer patients. Oncol Rev. 2010;4(4):223–32.
- Ricart AD. Antibody-drug conjugates of calicheamicin derivative: gemtuzumab ozogamicin and inotuzumab ozogamicin. Clin Cancer Res. 2011;17(20):6417–27.
- 22. Tallman MS, McDonald GB, DeLeve LD, Baer MR, Cook MN, Graepel GJ, et al. Incidence of sinusoidal obstruction syndrome following Mylotarg (gemtuzumab ozogamicin): a prospective observational study of 482 patients in routine clinical practice. Int J Hematol. 2013;97(4):456–64.
- 23. Jen EY, Ko CW, Lee JE, Del Valle PL, Aydanian A, Jewell C, et al. FDA approval: gemtuzumab ozogamicin for the treatment of adults with newly diagnosed cd33-positive acute myeloid leukemia. Clin Cancer Res. 2018;24(14):3242–6.
- 24. Lambert J, Pautas C, Terre C, Raffoux E, Turlure P, Caillot D, et al. Gemtuzumab ozogamicin for de novo acute myeloid leukemia: final efficacy and safety updates from the open-label, phase 3 ALFA-0701 trial. Haematologica. 2019;104:113–9.
- 25. McDonald GB, Freston JW, Boyer JL, DeLeve LD. Liver complications following treatment of hematologic malignancy with anti-cd22-calicheamicin (inotuzumab ozogamicin). Hepatology. 2019;69(2):831–44.
- 26. Kantarjian HM, DeAngelo DJ, Advani AS, Stelljes M, Kebriaei P, Cassaday RD, et al. Hepatic adverse event profile of inotuzumab ozogamicin in adult patients with relapsed or refractory acute lymphoblastic leukaemia: results from the open-label, randomised, phase 3 INO-VATE study. Lancet Haematol. 2017;4(8):e387–e98.
- 27. Chao N. How I treat sinusoidal obstruction syndrome. Blood. 2014;123(26):4023–6.
- 28. Al Beihany A, Al Omar H, Sahovic E, Chaudhri N, Al Mohareb F, Al Sharif F, et al. Successful treatment of hepatic veno-occlusive disease after myeloablative allogeneic hematopoietic stem cell transplantation by early administration of a short course of methylprednisolone. Bone Marrow Transplant. 2007;41:287.
- Seymour JF, Kipps TJ, Eichhorst B, Hillmen P, D'Rozario J, Assouline S, et al. Venetoclax–rituximab in relapsed or refractory chronic lymphocytic leukemia. N Engl J Med. 2018;378(12):1107–20.
- June CH, Warshauer JT, Bluestone JA. Is autoimmunity the Achilles' heel of cancer immunotherapy? Nat Med. 2017;23(5):540–7.
- Mehanna H, Paleri V, West CM, Nutting C. Head and neck cancer-part 1: epidemiology, presentation, and preservation. Clin Otolaryngol. 2011;36(1):65–8.
- June CH, Sadelain M. Chimeric antigen receptor therapy. N Engl J Med. 2018;379(1):64–73.
- Ling DC, Bakkenist CJ, Ferris RL, Clump DA. Role of immunotherapy in head and neck cancer. Semin Radiat Oncol. 2018;28(1):12–6.
- Szturz P, Vermorken JB. Immunotherapy in head and neck cancer: aiming at EXTREME precision. BMC Med. 2017;15(1):110.
- 35. Logan RM. Advances in understanding of toxicities of treatment for head and neck cancer. Oral Oncol. 2009;45(10):844–8.
- 36. Bertrand A, Kostine M, Barnetche T, Truchetet ME, Schaeverbeke T. Immune related adverse events associated with anti-CTLA-4 antibodies: systematic review and meta-analysis. BMC Med. 2015;13:211.
- 37. Thompson JA, Schneider BJ, Brahmer J, Andrews S, Armand P, Bhatia S, et al. NCCN Guidelines Insights: Management of Immunotherapy-Related Toxicities, Version 1.2020. J Natl Compr Canc Netw. 2020;18(3):230–41.
- 38. Brahmer JR, Lacchetti C, Schneider BJ, Atkins MB, Brassil KJ, Caterino JM, et al. Management of immune-related adverse events in patients treated with immune checkpoint inhibitor therapy: American Society of Clinical Oncology clinical practice guideline. J Clin Oncol. 2018;36(17):1714–68.
- Puzanov I, Diab A, Abdallah K, Bingham CO, Brogdon C, Dadu R, et al. Managing toxicities associated with immune checkpoint

inhibitors: consensus recommendations from the Society for Immunotherapy of Cancer (SITC) toxicity management working group. J Immuno Ther Cancer. 2017;5(1):95.

- Haanen JBAG, Carbonnel F, Robert C, Kerr KM, Peters S, Larkin J, Jordan K; ESMO Guidelines Committee. Management of toxicities from immunotherapy: ESMO clinical practice guidelines for diagnosis, treatment and follow-up. Ann Oncol. 2017;28(Suppl 4):iv119–42. Erratum in: Ann Oncol. 2018;29(Suppl 4):iv264–66. Erratum in: Ann Oncol. 2018;29 Suppl 4:iv264–66.
- 41. Abdel-Wahab N, Shah M, Suarez-Almazor ME. Adverse events associated with immune checkpoint blockade in patients with cancer: a systematic review of case reports. PLoS One. 2016;11(7):e0160221.
- 42. Maamari J, Yeung SJ, Chaftari PS. Diabetic ketoacidosis induced by a single dose of pembrolizumab. Am J Emerg Med. 2019;37(2):376 e1–2.
- Nishino M, Sholl LM, Hodi FS, Hatabu H, Ramaiya NH. Anti-PD-1-related pneumonitis during cancer immunotherapy. N Engl J Med. 2015;373(3):288–90.
- 44. Cappelli LC, Gutierrez AK, Baer AN, Albayda J, Manno RL, Haque U, et al. Inflammatory arthritis and sicca syndrome induced by nivolumab and ipilimumab. Ann Rheum Dis. 2017;76(1):43–50.
- 45. Cuzzubbo S, Javeri F, Tissier M, Roumi A, Barlog C, Doridam J, et al. Neurological adverse events associated with immune checkpoint inhibitors: review of the literature. Eur J Cancer. 2017;73:1–8.
- Jain V, Bahia J, Mohebtash M, Barac A. Cardiovascular complications associated with novel cancer immunotherapies. Curr Options Treat Cardiovasc Med. 2017;19(5):36.
- 47. Lankes K, Hundorfean G, Harrer T, Pommer AJ, Agaimy A, Angelovska I, et al. Anti-TNF-refractory colitis after checkpoint inhibitor therapy: possible role of CMV-mediated immunopathogenesis. Onco Targets Ther. 2016;5(6):e1128611.
- Cheng R, Cooper A, Kench J, Watson G, Bye W, McNeil C, et al. Ipilimumab-induced toxicities and the gastroenterologist. J Gastroenterol Hepatol. 2015;30(4):657–66.
- Weber JS, Hodi FS, Wolchok JD, Topalian SL, Schadendorf D, Larkin J, et al. Safety profile of nivolumab monotherapy: a pooled analysis of patients with advanced melanoma. J Clin Oncol. 2017;35(7):785–92.
- 50. Horvat TZ, Adel NG, Dang T-O, Momtaz P, Postow MA, Callahan MK, et al. Immune-related adverse events, need for systemic immunosuppression, and effects on survival and time to treatment failure in patients with melanoma treated with ipilimumab at Memorial Sloan Kettering Cancer Center. J Clin Oncol. 2015;33(28):3193.
- Johnson DB, Balko JM, Compton ML, Chalkias S, Gorham J, Xu Y, et al. Fulminant myocarditis with combination immune checkpoint blockade. N Engl J Med. 2016;375(18):1749–55.
- Mahmood SS, Fradley MG, Cohen JV, Nohria A, Reynolds KL, Heinzerling LM, et al. Myocarditis in patients treated with immune checkpoint inhibitors. J Am Coll Cardiol. 2018;71(16):1755–64.
- Escudier M, Cautela J, Malissen N, Ancedy Y, Orabona M, Pinto J, et al. Clinical features, management, and outcomes of immune checkpoint inhibitor-related cardiotoxicity. Circulation. 2017;136(21):2085–7.
- Wang DY, Okoye GD, Neilan TG, Johnson DB, Moslehi JJ. Cardiovascular toxicities associated with cancer immunotherapies. Curr Cardiol Rep. 2017;19(3):21.
- 55. Berg DD, Vaduganathan M, Nohria A, Davids MS, Alyea EP, Torre M, et al. Immune-related fulminant myocarditis in a patient receiving ipilimumab therapy for relapsed chronic myelomonocytic leukaemia. Eur J Heart Fail. 2017;19(5):682–5.
- 56. Behling J, Kaes J, Munzel T, Grabbe S, Loquai C. New-onset third-degree atrioventricular block because of autoimmuneinduced myositis under treatment with anti-programmed cell death-1 (nivolumab) for metastatic melanoma. Melanoma Res. 2017;27(2):155–8.

- Fellner A, Makranz C, Lotem M, Bokstein F, Taliansky A, Rosenberg S, et al. Neurologic complications of immune checkpoint inhibitors. J Neuro-Oncol. 2018;137(3):601–9.
- Zeng MF, Chen LL, Ye HY, Gong W, Zhou LN, Li YM, Zhao XL. Primary hypothyroidism and isolated ACTH deficiency induced by nivolumab therapy: case report and review. Medicine (Baltimore). 2017;96(44):e8426. Erratum in: Medicine (Baltimore). 2018;97(11):e0159.
- 59. Narita T, Oiso N, Taketomo Y, Okahashi K, Yamauchi K, Sato M, et al. Serological aggravation of autoimmune thyroid disease in two cases receiving nivolumab. J Dermatol. 2016;43(2):210–4.
- Tanaka R, Fujisawa Y, Maruyama H, Nakamura Y, Yoshino K, Ohtsuka M, et al. Nivolumab-induced thyroid dysfunction. Jpn J Clin Oncol. 2016;46(6):575–9.
- Diamantopoulos PT, Gaggadi M, Kassi E, Benopoulou O, Anastasopoulou A, Gogas H. Late-onset nivolumab-mediated pneumonitis in a patient with melanoma and multiple immunerelated adverse events. Melanoma Res. 2017;27(4):391–5.
- 62. Nandavaram S, Nadkarni A. Ipilimumab-induced sarcoidosis and thyroiditis. Am J Ther. 2018;25(3):e379–e80.
- Imafuku K, Yoshino K, Yamaguchi K, Tsuboi S, Ohara K, Hata H. Hypothyroidism associated with nivolumab treatment of unresectable malignant melanoma. Clin Exp Dermatol. 2017;42(2):217–8.
- 64. Yu C, Chopra IJ, Ha E. A novel melanoma therapy stirs up a storm: ipilimumab-induced thyrotoxicosis. Endocrinol Diabetes Metab Case Rep. 2015;2015:140092.
- 65. Khan U, Rizvi H, Sano D, Chiu J, Hadid T. Nivolumab induced myxedema crisis. J Immunother Cancer. 2017;5(1):13.
- 66. Hansen E, Sahasrabudhe D, Sievert L. A case report of insulindependent diabetes as immune-related toxicity of pembrolizumab: presentation, management and outcome. Cancer Immunol Immunother. 2016;65(6):765–7.
- 67. El Majzoub I, Qdaisat A, Thein KZ, Win MA, Han MM, Jacobson K, et al. Adverse effects of immune checkpoint therapy in cancer patients visiting the emergency department of a comprehensive cancer center. Ann Emerg Med. 2019;73(1):79–87.
- Torino F, Barnabei A, De Vecchis L, Salvatori R, Corsello SM. Hypophysitis induced by monoclonal antibodies to cytotoxic T lymphocyte antigen 4: challenges from a new cause of a rare disease. Oncologist. 2012;17(4):525–35.
- Corsello SM, Barnabei A, Marchetti P, De Vecchis L, Salvatori R, Torino F. Endocrine side effects induced by immune checkpoint inhibitors. J Clin Endrocrinol Metabol. 2013;98(4):1361–75.
- Ryder M, Callahan M, Postow MA, Wolchok J, Fagin JA. Endocrinerelated adverse events following ipilimumab in patients with advanced melanoma: a comprehensive retrospective review from a single institution. Endocr Relat Cancer. 2014;21(2):371–81.
- 71. Faje AT, Sullivan R, Lawrence D, Tritos NA, Fadden R, Klibanski A, et al. Ipilimumab-induced hypophysitis: a detailed longitudinal analysis in a large cohort of patients with metastatic melanoma. J Clin Endocrinol Metab. 2014;99(11):4078–85.
- 72. Tanaka F, Kogue R, Maeda M, Umino M, Nakai Y, Takeshita A, et al. Ipilimumab-induced hypophysitis involving the optic tracts and tuber cinereum evaluated using 3D fluid-attenuated inversion recovery. Radiol Case Rep. 2017;13(1):261–4.
- Miller AH, Brock P, Yeung S-CJ. Pituitary dysfunction: a case series of immune checkpoint inhibitor–related hypophysitis in an emergency department. Ann Emerg Med. 2016;68(2): 249–50.
- Postow MA. Managing immune checkpoint-blocking antibody side effects. Am Soc Clin Oncol Educ Book. 2015;35(76):76–83.
- 75. Faje A. Immunotherapy and hypophysitis: clinical presentation, treatment, and biologic insights. Pituitary. 2016;19(1):82–92.
- Min L, Ibrahim N. Ipilimumab-induced autoimmune adrenalitis. Lancet Diabetes Endocrinol. 2013;1(3):e15.

- 77. Trainer H, Hulse P, Higham CE, Trainer P, Lorigan P. Hyponatraemia secondary to nivolumab-induced primary adrenal failure. Endocrinol Diabetes Metab Case Rep. 2016;2016:16–0108.
- de Filette J, Jansen Y, Schreuer M, Everaert H, Velkeniers B, Neyns B, et al. Incidence of thyroid-related adverse events in melanoma patients treated with pembrolizumab. J Clin Endocrinol Metab. 2016;101(11):4431–9.
- 79. Morganstein D, Lai Z, Spain L, Diem S, Levine D, Mace C, et al. Thyroid abnormalities following the use of cytotoxic T-lymphocyte antigen-4 and programmed death receptor protein-1 inhibitors in the treatment of melanoma. Clin Endocrinol. 2017;86(4):614–20.
- McMillen B, Dhillon MS, Yong-Yow S. A rare case of thyroid storm. BMJ Case Rep. 2016;2016:bcr2016214603.
- Carroll R, Matfin G. Endocrine and metabolic emergencies: thyroid storm. Ther Adv Endocrinol Metab. 2010;1(3):139–45.
- 82. Sznol M, Postow MA, Davies MJ, Pavlick AC, Plimack ER, Shaheen M, et al. Endocrine-related adverse events associated with immune checkpoint blockade and expert insights on their management. Cancer Treat Rev. 2017;58:70–6.
- 83. Girotra M, Hansen A, Farooki A, Byun DJ, Min L, Creelan BC, et al. Investigational Drug Steering Committee (IDSC) Immunotherapy Task Force collaboration. The current understanding of the endocrine effects from immune checkpoint inhibitors and recommendations for management. JNCI Cancer Spectr. 2018;2(3):pky021.
- Hong AR, Yoon JH, Kim HK, Kang HC. Immune checkpoint inhibitor-induced diabetic ketoacidosis: a report of four cases and literature review. Front Endocrinol. 2020;11:14.
- 85. de Filette JMK, Pen JJ, Decoster L, Vissers T, Bravenboer B, Van der Auwera BJ, et al. Immune checkpoint inhibitors and type 1 diabetes mellitus: a case report and systematic review. Eur J Endocrinol. 2019;181(3):363–74.
- Akturk HK, Kahramangil D, Sarwal A, Hoffecker L, Murad MH, Michels AW. Immune checkpoint inhibitor-induced Type 1 diabetes: a systematic review and meta-analysis. Diabet Med. 2019;36(9):1075–81.
- Knight T, Ahn S, Rice TW, Cooksley T. Acute oncology care: a narrative review of the acute management of neutropenic sepsis and immune-related toxicities of checkpoint inhibitors. Eur J Int Med. 2017;45:59–65.
- Wang DY, Ye F, Zhao S, Johnson DB. Incidence of immune checkpoint inhibitor-related colitis in solid tumor patients: a systematic review and meta-analysis. Onco Targets Ther. 2017;6(10): e1344805.
- 89. Weber JS, Dummer R, de Pril V, Lebbe C, Hodi FS. Patterns of onset and resolution of immune-related adverse events of special interest with ipilimumab: detailed safety analysis from a phase 3 trial in patients with advanced melanoma. Cancer. 2013;119(9): 1675–82.
- Wang Y, Abu-Sbeih H, Mao E, Ali N, Qiao W, Trinh VA, et al. Endoscopic and histologic features of immune checkpoint inhibitorrelated colitis. Inflamm Bowel Dis. 2018;24(8):1695–705.
- Jain A, Lipson EJ, Sharfman WH, Brant SR, Lazarev MG. Colonic ulcerations may predict steroid-refractory course in patients with ipilimumab-mediated enterocolitis. World J Gastroenterl. 2017;23(11):2023–8.
- Pages C, Gornet JM, Monsel G, Allez M, Bertheau P, Bagot M, et al. Ipilimumab-induced acute severe colitis treated by infliximab. Melanoma Res. 2013;23(3):227–30.
- Merrill SP, Reynolds P, Kalra A, Biehl J, Vandivier RW, Mueller SW. Early administration of infliximab for severe ipilimumabrelated diarrhea in a critically ill patient. Ann Pharmacother. 2014;48(6):806–10.
- 94. Bergqvist V, Hertervig E, Gedeon P, Kopljar M, Griph H, Kinhult S, et al. Vedolizumab treatment for immune checkpoint inhibitor-induced enterocolitis. Cancer Immunol Immunother. 2017;66(5):581–92.

- 95. Diana P, Mankongpaisarnrung C, Atkins MB, Zeck JC, Charabaty A. Emerging role of vedolizumab in managing refractory immune checkpoint inhibitor-induced enteritis. ACG Case Rep J. 2018;5:e17.
- 96. Boutros C, Tarhini A, Routier E, Lambotte O, Ladurie FL, Carbonnel F, et al. Safety profiles of anti-CTLA-4 and anti-PD-1 antibodies alone and in combination. Nat Rev Clin Oncol. 2016;13(8):473–86.
- 97. Weitzman S. Approach to hemophagocytic syndromes. Hematology Am Soc Hematol Educ Program. 2011;2011:178–83.
- Ramos-Casals M, Brito-Zerón P, López-Guillermo A, Khamashta MA, Bosch X. Adult haemophagocytic syndrome. Lancet. 2014;383(9927):1503–16.
- Sadaat M, Jang S. Hemophagocytic lymphohistiocytosis with immunotherapy: brief review and case report. J Immunother Cancer. 2018;6(1):49.
- Shah D, Shrestha R, Ramlal R, Hatton J, Saeed H. Pembrolizumab associated hemophagocytic lymphohistiocytosis. Ann Oncol. 2017;28(6):1403.
- 101. Satzger I, Ivanyi P, Langer F, Kreipe HH, Schaper-Gerhardt K, Beutel G, et al. Treatment-related hemophagocytic lymphohistiocytosis secondary to checkpoint inhibition with nivolumab plus ipilimumab. Eur J Cancer. 2018;93:150–3.
- 102. Mobini N, Dhillon R, Dickey J, Spoon J, Sadrolashrafi K. Exclusive cutaneous and subcutaneous sarcoidal granulomatous inflammation due to immune checkpoint inhibitors: report of two cases with unusual manifestations and review of the literature. Case Rep Dermatol Med. 2019;2019:6702870.
- 103. Tetzlaff MT, Jazaeri AA, Torres-Cabala CA, Korivi BR, Landon GA, Nagarajan P, et al. Erythema nodosum-like panniculitis mimicking disease recurrence: a novel toxicity from immune checkpoint blockade therapy-report of 2 patients. J Cutan Pathol. 2017;44(12):1080–6.
- 104. Tetzlaff MT, Nelson KC, Diab A, Staerkel GA, Nagarajan P, Torres-Cabala CA, et al. Granulomatous/sarcoid-like lesions associated with checkpoint inhibitors: a marker of therapy response in a subset of melanoma patients. J Immunother Cancer. 2018; 6(1):14.
- 105. Belum VR, Benhuri B, Postow MA, Hellmann MD, Lesokhin AM, Segal NH, et al. Characterisation and management of dermatologic adverse events to agents targeting the PD-1 receptor. Eur J Cancer. 2016;60:12–25.
- Villadolid J, Amin A. Immune checkpoint inhibitors in clinical practice: update on management of immune-related toxicities. Transl Lung Cancer Res. 2015;4(5):560–75.
- 107. De Velasco G, Je Y, Bosse D, Awad MM, Ott PA, Moreira RB, et al. Comprehensive meta-analysis of key immune-related adverse events from ctla-4 and pd-1/pd-11 inhibitors in cancer patients. Cancer Immunol Res. 2017;5(4):312–8.
- Lacouture ME, Wolchok JD, Yosipovitch G, Kahler KC, Busam KJ, Hauschild A. Ipilimumab in patients with cancer and the management of dermatologic adverse events. J Am Acad Dermatol. 2014;71(1):161–9.
- Weber JS, Kahler KC, Hauschild A. Management of immunerelated adverse events and kinetics of response with ipilimumab. J Clin Oncol. 2012;30(21):2691–7.
- 110. Naidoo J, Page DB, Li BT, Connell LC, Schindler K, Lacouture ME, et al. Toxicities of the anti-PD-1 and anti-PD-L1 immune checkpoint antibodies. Ann Oncol. 2015;26(12):2375–91.
- Sibaud V. Dermatologic reactions to immune checkpoint inhibitors: skin toxicities and immunotherapy. Am J Clin Dermatol. 2018;19(3):345–61.
- 112. Kumar V, Chaudhary N, Garg M, Floudas CS, Soni P, Chandra AB. Current diagnosis and management of immune related

adverse events (iraes) induced by immune checkpoint inhibitor therapy. Front Pharmacol. 2017;8:49.

- 113. Naidoo J, Page DB, Li BT, Connell LC, Schindler K, Lacouture ME, et al. Toxicities of the anti-PD-1 and anti-PD-L1 immune checkpoint antibodies. Ann Oncol. 2016;27(7):1362.
- 114. Jaber SH, Cowen EW, Haworth LR, Booher SL, Berman DM, Rosenberg SA, et al. Skin reactions in a subset of patients with stage IV melanoma treated with anti-cytotoxic T-lymphocyte antigen 4 monoclonal antibody as a single agent. Arch Dermatol. 2006;142(2):166–72.
- 115. Voskens CJ, Goldinger SM, Loquai C, Robert C, Kaehler KC, Berking C, et al. The price of tumor control: an analysis of rare side effects of anti-CTLA-4 therapy in metastatic melanoma from the ipilimumab network. PLoS One. 2013;8(1):e53745.
- 116. Larkin J, Chmielowski B, Lao CD, Hodi FS, Sharfman W, Weber J, et al. Neurologic serious adverse events associated with nivolumab plus ipilimumab or nivolumab alone in advanced melanoma, including a case series of encephalitis. Oncologist. 2017;22(6):709–18.
- 117. Robert C, Schachter J, Long GV, Arance A, Grob JJ, Mortier L, et al. Pembrolizumab versus ipilimumab in advanced melanoma. N Engl J Med. 2015;372(26):2521–32.
- 118. Hodi FS, O'Day SJ, McDermott DF, Weber RW, Sosman JA, Haanen JB, et al. Improved survival with ipilimumab in patients with metastatic melanoma. N Engl J Med. 2010;363(8):711–23.
- 119. Eggermont AMM, Chiarion-Sileni V, Grob J-J, Dummer R, Wolchok JD, Schmidt H, et al. Prolonged survival in stage III melanoma with ipilimumab adjuvant therapy. N Engl J Med. 2016;375(19):1845–55.
- 120. Larkin J, Chiarion-Sileni V, Gonzalez R, Grob JJ, Cowey CL, Lao CD, et al. Combined nivolumab and ipilimumab or monotherapy in untreated melanoma. N Engl J Med. 2015;373(1):23–34.
- 121. Yang JC, Hughes M, Kammula U, Royal R, Sherry RM, Topalian SL, et al. Ipilimumab (anti-CTLA4 antibody) causes regression of metastatic renal cell cancer associated with enteritis and hypophysitis. J Immunother. 2007;30(8):825–30.
- 122. Bot I, Blank CU, Boogerd W, Brandsma D. Neurological immune-related adverse events of ipilimumab. Pract Neurol. 2013;13(4):278–80.
- 123. Oishi K, Nakao M, Maeda S, Matsushita T, Ikeda T, Yamada T, et al. A case of aseptic meningitis without neck rigidity occurring in a metastatic melanoma patient treated with ipilimumab. Eur J Dermatol. 2017;27(2):193–4.
- 124. Mandel JJ, Olar A, Aldape KD, Tremont-Lukats IW. Lambrolizumab induced central nervous system (CNS) toxicity. J Neurol Sci. 2014;344(1–2):229–31.
- 125. Salam S, Lavin T, Turan A. Limbic encephalitis following immunotherapy against metastatic malignant melanoma. BMJ Case Rep. 2016;2016:bcr2016215012.
- 126. Schneider S, Potthast S, Komminoth P, Schwegler G, Bohm S. PD-1 checkpoint inhibitor associated autoimmune encephalitis. Case Rep Oncol. 2017;10(2):473–8.
- 127. Stein MK, Summers BB, Wong CA, Box HL, Cleveland KO. Meningoencephalitis following ipilimumab administration in metastatic melanoma. Am J Med Sci. 2015;350(6):512–3.
- 128. Nishino M, Ramaiya NH, Awad MM, Sholl LM, Maattala JA, Taibi M, et al. PD-1 inhibitor–related pneumonitis in advanced cancer patients: radiographic patterns and clinical course. Clin Cancer Res. 2016;22(24):6051.
- 129. Cheng G-S, Possick JD, editors. Pulmonary complications of non-pulmonary malignancy. Clin Chest Med. 2017;38(2):A1-A10;17–384.
- 130. Chen TW, Razak AR, Bedard PL, Siu LL, Hansen AR. A systematic review of immune-related adverse event reporting in

clinical trials of immune checkpoint inhibitors. Ann Oncol. 2015;26(9):1824-9.

- 131. Topalian SL, Hodi FS, Brahmer JR, Gettinger SN, Smith DC, McDermott DF, et al. Safety, activity, and immune correlates of anti–PD-1 antibody in cancer. N Engl J Med. 2012;366(26):2443–54.
- 132. Garon EB, Rizvi NA, Hui R, Leighl N, Balmanoukian AS, Eder JP, et al. Pembrolizumab for the treatment of non–small-cell lung cancer. N Engl J Med. 2015;372(21):2018–28.
- 133. Gettinger SN, Horn L, Gandhi L, Spigel DR, Antonia SJ, Rizvi NA, et al. Overall survival and long-term safety of nivolumab (anti–programmed death 1 antibody, BMS-936558, ONO-4538) in patients with previously treated advanced non–small-cell lung cancer. J Clin Oncol. 2015;33(18):2004–12.
- 134. Lemiale V, Meert A-P, Vincent F, Darmon M, Bauer PR, Van de Louw A, et al. Severe toxicity from checkpoint protein inhibitors: what intensive care physicians need to know? Ann Intensive Care. 2019;9(1):25.
- 135. Belkhir R, Burel SL, Dunogeant L, Marabelle A, Hollebecque A, Besse B, et al. Rheumatoid arthritis and polymyalgia rheumatica occurring after immune checkpoint inhibitor treatment. Ann Rheum Dis. 2017;76(10):1747–50.
- 136. Bilen MA, Subudhi SK, Gao J, Tannir NM, Tu SM, Sharma P. Acute rhabdomyolysis with severe polymyositis following ipilimumab-nivolumab treatment in a cancer patient with elevated anti-striated muscle antibody. J Immunother Cancer. 2016; 4:36.
- 137. Ali SS, Goddard AL, Luke JJ, Donahue H, Todd DJ, Werchniak A, et al. Drug-associated dermatomyositis following ipilimumab therapy: a novel immune-mediated adverse event associated with cytotoxic T-lymphocyte antigen 4 blockade. JAMA Dermatol. 2015;151(2):195–9.
- Yoshioka M, Kambe N, Yamamoto Y, Suehiro K, Matsue H. Case of respiratory discomfort due to myositis after administration of nivolumab. J Dermatol. 2015;42(10):1008–9.
- Hunter G, Voll C, Robinson CA. Autoimmune inflammatory myopathy after treatment with ipilimumab. Can J Neurol Sci. 2009;36(4):518–20.
- Liao B, Shroff S, Kamiya-Matsuoka C, Tummala S. Atypical neurological complications of ipilimumab therapy in patients with metastatic melanoma. Neuro-Oncology. 2014;16(4):589–93.
- 141. Min L, Hodi FS. Anti-PD1 following ipilimumab for mucosal melanoma: durable tumor response associated with severe hypothyroidism and rhabdomyolysis. Cancer Immunol Res. 2014;2(1):15–8.
- 142. Diamantopoulos PT, Tsatsou K, Benopoulou O, Anastasopoulou A, Gogas H. Inflammatory myopathy and axonal neuropathy in a patient with melanoma following pembrolizumab treatment. J Immunother. 2017;40(6):221–3.
- 143. Konoeda F, Suzuki S, Nishimoto Y, Hoshino H, Takagi M. A case of myasthenia gravis and myositis induced by nivolumab. Rinsho Shinkeigaku. 2017;57(7):373–7. (Article in Japanese).
- 144. Shirai T, Sano T, Kamijo F, Saito N, Miyake T, Kodaira M, et al. Acetylcholine receptor binding antibody-associated myasthenia gravis and rhabdomyolysis induced by nivolumab in a patient with melanoma. Jpn J Clin Oncol. 2016;46(1):86–8.
- 145. Kimura T, Fukushima S, Miyashita A, Aoi J, Jinnin M, Kosaka T, et al. Myasthenic crisis and polymyositis induced by one dose of nivolumab. Cancer Sci. 2016;107(7):1055–8.
- 146. Vallet H, Gaillet A, Weiss N, Vanhaecke C, Saheb S, Touitou V, et al. Pembrolizumab-induced necrotic myositis in a patient with metastatic melanoma. Ann Oncol. 2016;27(7):1352–3.
- 147. Haddox CL, Shenoy N, Shah KK, Kao JC, Jain S, Halfdanarson TR, et al. Pembrolizumab induced bulbar myopathy and respi-

ratory failure with necrotizing myositis of the diaphragm. Ann Oncol. 2017;28(3):673–5.

- 148. Shah M, Tayar JH, Abdel-Wahab N, Suarez-Almazor ME. Myositis as an adverse event of immune checkpoint blockade for cancer therapy. Semin Arthritis Rheum. 2019;48(4):736–40.
- 149. Calabrese C, Kirchner E, Kontzias K, Velcheti V, Calabrese LH. Rheumatic immune-related adverse events of checkpoint therapy for cancer: case series of a new nosological entity. RMD Open. 2017;3(1):e000412.
- 150. Lecouflet M, Verschoore M, Giard C, Gohier P, Le Corre Y, Milea D, Martin L. Orbital myositis associated with ipilimumab. Ann Dermatol Venereol. 2013;140(6–7):448–51. (Article in French).
- 151. Pushkarevskaya A, Neuberger U, Dimitrakopoulou-Strauss A, Enk A, Hassel JC. Severe ocular myositis after ipilimumab treatment for melanoma: a report of 2 cases. J Immunother. 2017;40(7):282–5.
- Fox E, Dabrow M, Ochsner GA. Case of nivolumab-induced myositis. Oncologist. 2016;21(12):e3.
- 153. Byun DJ, Wolchok JD, Rosenberg LM, Girotra M. Cancer immunotherapy-immune checkpoint blockade and associated endocrinopathies. Nat Rev Endocrinol. 2017;13(4):195–207.
- 154. Davila ML, Riviere I, Wang X, Bartido S, Park J, Curran K, et al. Efficacy and toxicity management of 19-28z CAR T cell therapy in B cell acute lymphoblastic leukemia. Sci Transl Med. 2014;6(224):224ra25.
- Lee DW, Gardner R, Porter DL, Louis CU, Ahmed N, Jensen M, Grupp SA, Mackall CL. Current concepts in the diagnosis and management of cytokine release syndrome. Blood. 2014;124(2):188– 95. Erratum in: Blood. 2015;126(8):1048. Dosage error in article text. Erratum in: Blood. 2016;128(11):1533.
- 156. Locke FL, Neelapu SS, Bartlett NL, Siddiqi T, Chavez JC, Hosing CM, et al. Phase 1 results of ZUMA-1: a multicenter study of KTE-C19 anti-CD19 CAR T cell therapy in refractory aggressive lymphoma. Mol Ther. 2017;25(1):285–95.
- 157. Maude SL, Barrett D, Teachey DT, Grupp SA. Managing cytokine release syndrome associated with novel T cell-engaging therapies. Cancer J. 2014;20(2):119–22.
- 158. Lee DW, Santomasso BD, Locke FL, Ghobadi A, Turtle CJ, Brudno JN, et al. ASTCT consensus grading for cytokine release syndrome and neurologic toxicity associated with immune effector cells. Biol Blood Marrow Transplant. 2019;25(4):625–38. https://doi.org/10.1016/j.bbmt.2018.12.758.
- Brudno JN, Kochenderfer JN. Recent advances in CAR T-cell toxicity: mechanisms, manifestations and management. Blood Rev. 2019;34:45–55. https://doi.org/10.1016/j.blre.2018.11.002.
- 160. Neelapu SS, Tummala S, Kebriaei P, Wierda W, Gutierrez C, Locke FL, et al. Chimeric antigen receptor T-cell therapy assessment and management of toxicities. Nat Rev Clin Oncol. 2018;15(1):47–62.
- Brudno JN, Kochenderfer JN. Toxicities of chimeric antigen receptor T cells: recognition and management. Blood. 2016;127(26):3321–30.
- 162. Kochenderfer JN, Dudley ME, Kassim SH, Somerville RP, Carpenter RO, Stetler-Stevenson M, et al. Chemotherapyrefractory diffuse large B-cell lymphoma and indolent B-cell malignancies can be effectively treated with autologous T cells expressing an anti-CD19 chimeric antigen receptor. J Clin Oncol. 2015;33(6):540.
- 163. Turtle CJ, Hanafi L-A, Berger C, Hudecek M, Pender B, Robinson E, et al. Immunotherapy of non-Hodgkin's lymphoma with a defined ratio of CD8+ and CD4+ CD19-specific chimeric antigen receptor–modified T cells. Sci Transl Med. 2016;8(355):355ra116.
- 164. Turtle CJ, Hay KA, Hanafi L-A, Li D, Cherian S, Chen X, et al. Durable molecular remissions in chronic lymphocytic leukemia

treated with CD19-specific chimeric antigen receptor-modified T cells after failure of ibrutinib. J Clin Oncol. 2017;35(26): 3010–20.

- 165. Kochenderfer JN, Somerville RP, Lu T, Shi V, Bot A, Rossi J, et al. Lymphoma remissions caused by anti-CD19 chimeric antigen receptor T cells are associated with high serum interleukin-15 levels. J Clin Oncol. 2017;35(16):1803–13.
- 166. Schuster SJ, Bishop MR, Tam CS, Waller EK, Borchmann P, McGuirk JP, et al. JULIET Investigators. Tisagenlecleucel in adult

relapsed or refractory diffuse large B-cell lymphoma. N Engl J Med. 2019;380(1):45–56.

- 167. Acharya UH, Dhawale T, Yun S, Jacobson CA, Chavez JC, Ramos JD, et al. Management of cytokine release syndrome and neuro-toxicity in chimeric antigen receptor (CAR) T cell therapy. Expert Rev Hematol. 2019;12(3):195–205.
- Yescarta [package insert]. Santa Monica: Kite Pharma, Inc; 2020. https://www.fda.gov/media/108377/download. Accessed 20 Oct 2020.



Fluorouracil or Capecitabine Overdose

Natalie R. Neumann and Christopher O. Hoyte

Case Study

A 62-year-old female recently diagnosed with metastatic colon cancer is brought to the emergency department (ED) by family members after revealing to them that she ingested her chemotherapeutic medication in a self-harm attempt. She reports to ED staff that she took several weeks' worth of her prescribed capecitabine. The ingestion was 8 hours ago, and the patient is currently asymptomatic. Her physical exam, EKG, and laboratory tests including a complete metabolic panel, complete blood count, and troponin are all reassuring. After discussing the case with the patient's oncologist and a toxicologist through the local poison center, the patient is administered uridine triacetate and admitted to the hospital for further observation and serial laboratory assessment.

History and Epidemiology

5-Fluorouracil (5-FU) is a mainstay of oncologic therapy. Its development is a classic example of rational drug design; it was created in the 1950s when researchers discovered that malignant cells utilized exogenous uracil, a pyrimidine base, more rapidly than nonmalignant cells to sustain ribonucleic acid (RNA) synthesis and thus tumor growth [1–4]. It is classified as a fluoropyrimidine derivative, its structure consist-

N. R. Neumann (🖂)

C. O. Hoyte

ing of a pyrimidine bonded to a fluorine atom. 5-FU is, as the name suggests, a uracil analogue bonded to a fluorine atom in place of a hydrogen atom [3]. Other subsequently developed fluoropyrimidine derivatives that are used in oncologic therapy include capecitabine, doxifluridine, floxuridine, carmofur, and tegafur [5-9]. Fluoropyrimidines are themselves classified as antimetabolic agents. Antimetabolites are compounds that are similar enough to natural chemicals that they participate in physiologic biosynthetic pathways, but different enough that they also disrupt normal cell functions [5]. For the sake of simplicity, this chapter will focus largely on 5-FU and its prodrug, capecitabine. Cytotoxicity from these two fluoropyrimidines is attributed to their deleterious incorporation into RNA and deoxyribonucleic acid (DNA), and to their inhibition of thymidylate synthetase [1, 3, 4, 10] (Fig. 54.1).

Fluoropyrimidines may be administered as monotherapy or in combination with other chemotherapeutic agents. They are predominantly used for the treatment of gastrointestinal, breast, and head and neck cancers [1, 11, 12]. As of 2008, several hundred thousand patients in the United States received either 5-FU or capecitabine annually [13]. 5-FU is an intra-



Fig. 54.1 Capecitabine and 5-fluorouracil chemical structures. (Adapted from Walko et al. [10], with permission Elsevier)

Note: Dr. Neumann performed this research at the Rocky Mountain Poison and Drug Safety, Denver Health and Hospital Authority, Denver, Colorado. She is now with the Department of Emergency Medicine, Yale University School of Medicine, New Haven, Connecticut.

Department of Emergency Medicine, Yale University School of Medicine, New Haven, CT, USA

Rocky Mountain Poison and Drug Safety, Denver Health and Hospital Authority, Denver, CO, USA

Department of Emergency Medicine, University of Colorado School of Medicine, Aurora, CO, USA

K. H. Todd et al. (eds.), Oncologic Emergency Medicine, https://doi.org/10.1007/978-3-030-67123-5_54



Fig. 54.2 Diagram of the conversion of the prodrug, capecitabine, to the active drug, 5-fluorouracil (5-FU). Capecitabine is first absorbed through the gastrointestinal tract and then converted by carboxylesterase in the liver to 5'-deoxy-5-fluorocytidine (5'-DFCR). Cytidine deaminase, present in both normal (e.g., liver) and malignant tissues, subsequently metabolizes 5'-DFCR to 5'-deoxy-5-fluorouridine (5'-DFUR). Thymidine phosphorylase, generally found in higher concentrations in tumor tissue compared to normal tissue, will finally convert 5'-DFUR to 5-FU. (Adapted from Walko et al. [10], with permission Elsevier)

venous (IV) formulation approved by the Food and Drug Administration (FDA) for the treatment of adenocarcinoma of the colon and rectum, breast, stomach, and pancreas [12]. It may be administered as a bolus or infusion [12]. Capecitabine is an oral prodrug which is metabolized intracellularly to 5-FU (Fig. 54.2) [10]. It was designed with the goal of creating a medication that delivered not only higher intra-tumor concentrations but greater convenience and tolerability for patients (i.e., fewer adverse effects) [10, 11, 14, 15]. According to the FDA, capecitabine is indicated for the treatment of metastatic colorectal cancer and metastatic breast cancer, and as an adjuvant treatment for colon cancer [11]. Survival and tumor response rates for both medications vary depending on the type of cancer being treated. Similarly, while adverse reactions to therapeutic administration of 5-FU and capecitabine are quite common, the exact rates vary depending on patient age, cancer type, means of administration, and whether the agents are used as monotherapy [11, 14, 16]. Studies differ, but it is estimated that between 10% and 40% of patients on 5-FU and capecitabine develop some form of severe or life-threatening toxicity, including mucositis, myelosuppression, neurotoxicity, cardiotoxicity, and hand-and-foot syndrome [16-25]. Mortality from 5-FU and capecitabine-induced toxicity is estimated at approximately 0.4-2% [16, 24, 26, 27]. A significant proportion of fatalities are thought to occur in the setting of therapeutic administration rather than acute overdose, and they are often attributed to genetic polymorphisms [18, 22, 28–31].

Pharmacodynamics

5-FU, as noted previously, is an analogue of uracil and an antimetabolite chemotherapeutic agent. It is hypothesized that 5-FU is taken up into cells via the same active trans-

port system used to take up uracil [32]. Capecitabine, once ingested orally, is metabolized to 5-FU in liver and tumor cells. One enzyme critical to this conversion is cytidine deaminase (CDA) which is present in both normal and malignant tissues [11, 33, 34]. The third and final enzyme responsible for converting capecitabine to 5-FU is thymidine phosphorylase. Thymidine phosphorylase is thought to be found in much higher concentrations in tumor cells compared to normal tissues. Capecitabine's ability to localize to tumor cells and its improved side effect profile compared to 5-FU are attributed to this disparity in distribution [11, 34, 35].

Once 5-FU is present within the cell (after either direct uptake or metabolism of capecitabine), it is converted to three principal active metabolites which induce cytotoxicity by interfering with the function of thymidylate synthase (TS) and with the synthesis of RNA and DNA [1, 3]. These metabolites are fluorodeoxyuridine monophosphate (FdUMP), fluorouridine triphosphate (FUTP), and fluorodeoxyuridine triphosphate (FdUTP). Conversion of 5-FU to these active metabolites is accomplished in several steps and by multiple enzymes, including orotate phosphoribosyltransferase (OPRT) [1, 36]. Most administered 5-FU is not anabolized to an active metabolite, however. The enzyme dihydropyrimidine dehvdrogenase (DPD) catabolizes approximately 80% to an inactive component which is largely excreted in urine [4, 37]. The aforementioned enzymes - CDA, TS, OPRT, and DPD - are relevant to understanding the toxicity of 5-FU as their genetic polymorphisms are associated with variable risk of developing clinical toxicity at therapeutic fluoropyrimidine dosing (Fig. 54.3) [1, 18, 30].

Thymidylate synthase is an enzyme that catalyzes the conversion of deoxyuridine monophosphate (dUMP) to deoxythymidine monophosphate (dTMP), a nucleotide needed to synthesize DNA. Under physiologic conditions, thymidylate synthase uses 5,10-methylene tetrahydrofolate as a methyl donor to convert dUMP to dTMP. FdUMP, however, binds to TS with higher affinity than endogenous dUMP [38]. When present in the cell, FdUMP will form a stable but reversible ternary complex with thymidylate synthase (a dimer) and 5,10-methylene tetrahydrofolate [38–41]. Formation of this complex prevents binding of dUMP and so inhibits de novo dTMP production [1, 3]. This process is depicted in Fig. 54.4.

The exact means by which dTMP depletion leads to inhibition of DNA synthesis is not entirely known. Inadequate conversion of dUMP to dTMP is thought to create to a relative overabundance of dUTP. dUTP may then be incorporated inappropriately into DNA, resulting in strand damage and cell death [42–47]. FdUTP, another metabolite of 5-FU, can also be misincorporated into DNA, causing similar downstream effects [48–51]. In addition, decreased production of dTMP ultimately results in decreased dTTP availability which is thought to alter the concentrations of other



Fig. 54.3 A more detailed overview of the metabolism and mechanism of action of 5-FU. 5-FU is mostly catabolized by dihydropyrimidine dehydrogenase (DPD) to the inactive compound dihydrofluorouracil (DHFU). Much of this conversion takes place in the liver, but catabolism may also take place in other normal tissues and in tumor tissue. 5-FU may alternatively be metabolized to one of the three active metabolites: fluorodeoxyuridine monophosphate (FdUMP), fluorodeoxyuridine triphosphate (FdUTP), and fluorouridine triphosphate (FUTP). FUTP is formed in one of the two ways: 5-FU may be metabolized to fluorouridine monophosphate (FUMP) by orotate phosphoribosyltransferase (OPRT) with phosphoribosyl pyrophosphate (PRPP) acting as a cofactor. Alternatively, 5-FU may be metabolized by uridine phosphorylase (UP) to fluorouridine (FUR) and then to FUMP by uridine kinase (UK). FUMP, once formed, is phosphorylated to make fluorouridine diphosphate (FUDP) which may be subsequently phosphorylated to form FUTP. FUDP, rather than being converted to FUTP, can alternatively be metabolized to fluorodeoxyuridine diphosphate (FdUDP) by ribonucleotide reductase (RR). FdUDP may then be phosphorylated or dephosphorylated to form FdUTP or FdUMP, respectively. FdUMP is also formed if 5-FU, rather than being metabolized by UP or OPRT, is converted by thymidine phosphorylase (TP) to fluorodeoxyuridine (FUDR) and then to FdUMP by thymidylate kinase (TK). (From Longley et al. [1], with permission Springer Nature. © 2003 Nature Publishing Group)

deoxynucleotides. This imbalance is itself thought to damage DNA synthesis and repair [52, 53]. Finally, the abnormal ratio of dUTP to dTTP inhibits the action of uracil-DNA glycosylase, which as a nucleotide excision repair enzyme would normally remove inappropriate, uracil-based nucleotides from DNA strands [1, 48, 54]. Of note, thymidylate can be salvaged by thymidine kinase (TK). Increased intra-tumor thymidine kinase activity is a proposed mechanism of resistance to therapy (Fig. 54.4) [1, 55].

RNA disruption is achieved through multiple mechanisms as well, and is increasingly thought to be a significant contributor to cytotoxicity [47]. When present in the cell, FUTP, a 5-FU metabolite, may replace uridine triphosphate (UTP) and be incorporated into all forms of RNA [56, 57]. In this way it can disrupt the function of ribosomal RNA (rRNA), messenger RNA (mRNA), translational RNA (tRNA), and small nuclear RNA (snRNA) [3, 58–62]. Specifically, FUTP prevents the formation of mature rRNA by inhibiting prerRNA processing, disrupts the posttranslational modification of tRNA, inhibits polyadenylation of mRNA, and prevents snRNA-protein complex assembly [63–68]. Together these actions inhibit cellular function and may lead to cell death.

Pharmacokinetics

5-FU is administered intravenously. When given as a bolus, 5–20% is excreted unchanged in the urine over 6 hours. What remains is largely metabolized in normal and malignant tissues to active and inactive metabolites. The majority of 5-FU is catabolized (i.e., degraded to inactive compounds) by DPD in the liver and other tissues [3, 4, 12, 47, 69]. The degree to which 5-FU undergoes anabolism to metabolically active metabolites is thought to depend on the extent of catabolism [4, 37]. In other words, the more 5-FU is catabolized by DPD, the less is available to be converted to active metabolites. In general, less than 5% of 5-FU is anabolized [4].

Variable elimination half-lives and volumes of distribution have been reported for 5-FU; these range from 8 to 22 minutes and 14 to 54 liters, respectively [3, 12, 69–71]. The vast majority of 5-FU metabolites are excreted in the urine within 24 hours; approximately 2–3% undergo biliary excretion [37]. The half-lives of 5-FU's catabolites vary depending on the compound from 1 to 33 hours [3, 37]. Pharmacokinetic data regarding 5-FU continuous infusions is limited [3]. There are no specific recommendations for adjusting the dose of 5-FU in the setting of renal or hepatic dysfunction [3, 12, 72].

The bioavailability of IV 5-FU is by definition 100%. Tissue concentrations, however, vary significantly (up to a thousandfold difference) depending not only on the degree of hepatic metabolism but on whether or not the medication is administered as a bolus or as a slow infusion [71]. The oral bioavailability of 5-FU is highly variable and generally considered to be quite low given first-pass metabolism (one study reported a mean bioavailability of 28% with a range of 0–74%) [71, 73, 74]. Capecitabine also has rela-



Fig. 54.4 Diagram of 5-FU's inhibition of thymidylate synthase (TS). TS uses 5,10-methylene tetrahydrofolate (CH_2THF) to catalyze the conversion of deoxyuridine monophosphate (dUMP) to deoxythymidine monophosphate (dTMP). Fluorodeoxyuridine monophosphate (FdUMP), a 5-FU metabolite, interferes in this process by binding at the nucleotide-binding site of TS to form a stable ternary complex with TS and CH₂THF. When this complex forms, dUMP is unable to access

the nucleotide-binding site, and dTMP synthesis is inhibited. This inhibition leads to deoxynucleotide (dNTP) pool imbalances and increased concentrations of dUTP, and DNA damage ensues. The degree of DNA damage caused by dUTP depends on pyrophosphatase dUTPase (dUTPase) and uracil-DNA glycosylase (UDG) concentrations. Thymidine kinase (TK) may salvage dTMP from thymidine. (From Longley et al. [1], with permission Springer Nature. © 2003 Nature Publishing Group)

tively low bioavailability. A study measuring the serum presence of 5'-deoxy-5-fluorouridine (the immediate precursor to 5-FU in the metabolism of capecitabine) determined that capecitabine had a mean oral bioavailability of 42% in cancer patients with normal hepatic function and 62% in those with hepatic dysfunction [75].

After oral dosing, capecitabine reaches peak serum concentrations at 1.5 hours; peak 5-FU concentrations occur at 2 hours [11]. Food consumption decreases both the rate and absolute amount of capecitabine absorption and delays by 1.5 hours the time to peak serum concentrations of both capecitabine and 5-FU. Less than 60% of capecitabine and its metabolites are protein bound, and this percentage is not dependent on drug concentrations [11, 76]. As noted previously, capecitabine undergoes three enzymatic steps to produce 5-FU; subsequent metabolism is identical to that of 5-FU [10, 11]. Mild to moderate hepatic dysfunction is not shown to have a clinically significant impact on the pharmacokinetics of capecitabine and its metabolites, but the FDA recommends that patients with hepatic dysfunction be monitored carefully on the medication [11, 75]. Similarly, no dose adjustment is initially required in the setting of mild renal dysfunction, but capecitabine is contraindicated in the setting of renal failure as defined by a creatinine clearance of less than 30 mL/min. The FDA also recommends that the dose be decreased in the setting of moderate renal impairment as defined by a creatinine clearance of 30–50 mL/min [11].

Of note, both 5-FU and capecitabine are found to decrease warfarin clearance, leading to an increase in INR [11, 77]. They are also known to increase phenytoin concentrations [11, 78]. Both warfarin and phenytoin are metabolized by

CYP2C9 in the liver, suggesting that other drugs metabolized by this pathway should be closely monitored [10].

Toxicokinetics

Toxicity from 5-FU and capecitabine can result from intentional overdose, iatrogenic overdose, and therapeutic dosing [79–81]. Ma and colleagues, in one of the first studies exploring the use of uridine triacetate, noted that toxicity due to overdose was both rate and dose dependent. When three times the appropriate dose was administered, the observed outcome was death [28].

Toxicity in the setting of therapeutic drug administration is often associated with genetic variations that lead to abnormal enzyme activity [80, 81]. Genetic polymorphisms can lead to either excessive or deficient enzymatic activity. Polymorphisms associated with increased fluoropyrimidine toxicity are found in genes that code for the previously mentioned enzymes: DPD, CDA, and TS [18, 25, 29–31, 36, 82– 86]. Other enzymes that are suspected of causing increased toxicity due to genetic polymorphisms include methylene tetrahydrofolate reductase (MTHFR) and orotate phosphoribosyltransferase (OPRT) [25, 30, 36].

Of these polymorphisms, DPD is the best studied. It is generally accepted that a significant proportion of patients who present with severe 5-FU- or capecitabine-induced toxicity (defined previously in studies as symptoms consistent with grade 3 or 4 toxicity) have a partial or complete DPD deficiency. Approximately 25–60% of patients with severe toxicity have a DPD deficiency, and such findings have led to recommendations that genetic testing be performed prior to initiating fluoropyrimidine therapy [21, 30, 87–93]. Of note, it is estimated that only 2.7–5% of the general population (including those with cancer) has a partial DPD deficiency [93, 94]. Reduced DPD activity is not associated with increased age or abnormal liver function, but its relationship to gender is unclear, and rates of deficiency vary greatly between ethnicities [25, 93–95].

Risk factors that have been associated with higher rates of severe toxicity at therapeutic capecitabine dosing include increased age, pre-treatment with uracil, decreased renal function, decreased body surface area, female sex, and the use of capecitabine in combination therapy [24]. Abnormal hepatic function has not been clearly associated with increased risk of toxicity [33, 72, 75].

Data characterizing acute, one-time lethal doses of capecitabine and 5-FU in patients or subjects not undergoing treatment for malignancy is scarce. According to Burns and colleagues, Hoffmann-La Roche reported an LD50 in male Swiss mice of 365 mg/kg [96]. Johnson et al. reported a murine LD90 of 458 mg/kg [97]. A third study found that 50 mg/kg of 5-FU given once daily for 3 days resulted in 40% mortality in a mouse model while 60 mg/kg resulted in 100% [98]. Interestingly, the lethal dose is thought to vary based on circadian rhythms [96]. With respect to capecitabine specifically, the FDA reports that doses up to 2000 mg/kg are not lethal in mice, rats, or monkeys [11]. There is no data available regarding changes to bioavailability in the overdose setting for either medication.

Pathophysiology

The pathophysiology of toxic effects and the pharmacology of therapeutic effects are largely the same. Fluoropyrimidines are designed to kill cells; they preferentially destroy malignant cells, but in killing tumors, they may also damage normal tissue. There are, however, two unique toxicities that bear mentioning: hand-and-foot syndrome and fluoropyrimidineinduced cardiotoxicity.

Hand-and-foot syndrome, also known as palmar-plantar erythrodysesthesia and chemotherapy-induced acral erythema, is one of the few side effects seen more commonly in patients on capecitabine rather than 5-FU [11, 15, 99]. The mechanism is not entirely clear, but one popular theory suggests that higher concentrations of thymidylate phosphorylase in the palms as well as increased basal cell proliferation may be responsible [99–101]. Another study found a significant correlation between the presence of a particular methylene tetrahydrofolate reductase genotype and the development of hand-and-foot syndrome [30].

Reports describing the epidemiology of fluoropyrimidineinduced cardiotoxicity vary in their assessments. Most reports suggest a prevalence of around 4% [102–107]. In a review of 1350 patients without pre-existing cardiac disease, how-

ever, only 1.2% of patients were found to have cardiac complications [108]. Symptoms can range from asymptomatic EKG changes to myocardial infarction (MI), heart failure, and cardiogenic shock [104, 109]. Patients may also develop cardiomyopathy with left ventricular systolic dysfunction, ventricular dysrhythmia, arterial vasospasm, and direct endothelial damage. The pathogeneses of these phenomena are not entirely known [103, 108, 110-112]. Coronary vasospasm has historically been considered the most significant contributor to cardiotoxicity [108]. More recently, researchers have suggested that endothelial dysfunction, thrombus formation, direct myocardial damage, and accumulation of cytotoxic 5-FU metabolites may also play a role [108, 111]. The majority of case reports and review articles describing 5-FU- and capecitabine-induced toxicity describe symptoms occurring while patients are actively undergoing therapy [104, 105, 108–111, 113, 114]. Its occurrence appears to be dependent on the dose and means of administration; its relationship to pre-existing cardiovascular disease is unclear [102, 103, 107, 114].

Clinical Presentation

Toxicity from fluoropyrimidines may be considered as being on the spectrum of adverse effects due to chemotherapy. The Common Toxicity Score (CTS) developed by the National Cancer Institute (NCI) characterizes all symptoms of toxicity from mild adverse effects to death according to a five-point scale. Grade 0 describes absent symptoms, grade 1 mild, grade 2 moderate, grade 3 severe and undesirable, grade 4 life-threatening or disabling, and grade 5 death. The definition of mild, moderate, severe, etc. depends on the particular toxicity in question (e.g., neurologic versus hematologic) [115]. To the emergency physician, symptoms consistent with grades 3, 4, and 5 are the most concerning.

Common adverse effects from 5-FU and capecitabine include fatigue, nausea, vomiting, stomatitis, abdominal pain, hyperbilirubinemia, anemia, leukopenia, and neutropenia (neutropenia being the most common hematologic toxicity) [11, 26]. Peripheral sensory neuropathies may also occur, as well as nonspecific neurologic findings including headache, dizziness, and insomnia. Alopecia, venous thrombosis, and predisposition to infection are among other complications. The frequency of each adverse event and the proportion of patients who develop severe or life-threatening signs and symptoms depend on patient characteristics, the type of malignancy being treated, and the treatment protocol being used. (5-FU demonstrates different rates of adverse events when it is delivered as a bolus versus as a continuous infusion; rates also vary when fluoropyrimidines are administered as monotherapy versus combination therapy.) As discussed, capecitabine is generally better tolerated than

5-FU although it is associated with higher rates of hand-and-foot syndrome. Cassidy et al. reported stomatitis in 61.6% of patients with metastatic colon cancer on 5-FU with leucovorin while 24.3% developed stomatitis on capecitabine. In the same study, 10.3% of patients developed neutropenia on 5-FU versus 1.2% of those on capecitabine [1, 10–12, 14–16, 24, 26, 35, 72, 116].

Death and life-threatening toxicity are generally associated with neutropenia, leukopenia, severe diarrhea, and mucositis. A combination of these effects, especially neutropenic fever and mucositis or diarrhea, can lead to sepsis, shock, and multi-organ dysfunction [19, 24, 26, 80]. Death from therapy occurs in about 0.4–2% of patients, and severe toxicity frequently presents during the first treatment cycle [16, 24, 26, 27, 80]. Other rare but potentially fatal complications include central neurologic dysfunction and 5-FU-induced cardiotoxicity. Neurotoxicity may vary in presentation: ranging from confusion and altered mentation to frank encephalopathy and coma [80, 117–121].

The clinical manifestations of cardiovascular toxicity are described in the section on toxicokinetics. They range from asymptomatic EKG changes to cardiovascular collapse. The more dramatic presentations, including myocardial infarction, coronary dissection, ventricular dysrhythmias, cardiogenic shock, and heart failure, are likely to be obvious to the emergency physician. Death from cardiotoxicity results most frequently from heart failure and sudden cardiac death. Patients can also present, however, with anginal symptoms, asymptomatic ST segment changes, and QT interval prolongation [103, 106, 108, 112, 114, 122-124]. Rezkalla et al., in a prospective study, found that 68% of patients receiving 5-FU infusion had asymptomatic ST segment changes (versus 24% who had asymptomatic abnormalities prior to treatment). The study reported that while CK-MB concentrations generally increased during treatment, ST segments all returned to baseline by the end of the study and no patients developed myocardial infarction. Of note, however, two of the patients who developed ST segment changes and ventricular ectopy during therapy died within hours of completing the study; new infarctions were not noted on patient autopsies [109]. In another study, De Forni et al. found that 62.5% of patients who developed anginal symptoms during treatment demonstrated myocardial dysfunction on echocardiogram (ECHO). All ECHOs returned to normal within 2-6 days [123].

Hand-and-foot syndrome, a characteristic adverse effect of fluoropyrimidines, is generally graded according to a three-point severity score [11]. Symptoms range from paresthesias or painless swelling of the palms and soles (grade 1), to painful erythema and edema which do not interfere greatly with activities of daily living (grade 2), to obvious desquamation, blistering, ulceration, and pain which interfere with daily activities (grade 3) [11, 100, 101].

Evaluation and Diagnostic Testing

A diagnosis of fluoropyrimidine-induced toxicity is made by history, physical exam, and laboratory assessment. Laboratory tests may include a complete blood cell count and complete metabolic panel, including liver enzymes. If a patient complains of anginal symptoms, an EKG and troponin should be obtained, and an echocardiogram should be considered. Given the frequency with which patients develop silent ischemia (including ST segment changes without chest pain), physicians may consider performing an EKG on even those patients who do not present with anginal symptoms. If an EKG is performed and is found to be abnormal, consideration should be given to performing cardiac enzymes [109]. Other testing should be directed toward the patient's chief complaint (e.g., blood cultures for infection, lumbar puncture in the setting of acute altered mentation, etc.). There are no specialized laboratory tests that are required emergently, and there is no need to obtain plasma 5-FU concentrations. This test will not change management and it may take days to weeks to result.

Management

Management of patients presenting with fluoropyrimidineinduced toxicity consists first and foremost of stabilization, resuscitation, and supportive care. Consultation with the treating oncologist, if possible, is recommended. In many cases, toxicity can be managed with simple dose reduction [10, 11, 15, 30, 72, 88, 100, 113]. Uridine triacetate is the well-publicized and expensive antidote for fluoropyrimidineinduced toxicity. Its indications are limited, however, and ideally an emergency physician would discuss giving this medication with an oncologist and/or toxicologist prior to its administration. Uridine triacetate will be discussed in the section that follows.

With respect to cardiotoxicity specifically, stabilization and supportive care take priority. Given that most patients who develop cardiotoxicity are actively undergoing treatment with a fluoropyrimidine, one of the first steps consists of discontinuing therapy (i.e., turn off the infusion pump). Prophylaxis and treatment with calcium channel blockers and nitroglycerin have been studied, with inconsistent results [103, 106, 112, 125]. Given the conflicting data, it seems appropriate to consider these medications in consultation with a cardiologist, oncologist, or toxicologist, and, more importantly, to use conventional therapies (e.g., follow standard ACLS guidelines) in patients who are unstable [106, 108, 112]. There is insufficient evidence to recommend the use of steroids in the setting of cardiotoxicity [110]. Patients with anginal symptoms and elevated cardiac enzymes should be admitted to the hospital, their cardiac enzymes trended and consideration be given to performing an ECHO [108, 123]. The treating oncologist or cardiologist should be consulted regarding the management of patients undergoing therapy who present with asymptomatic EKG changes and without laboratory evidence of ischemia.

Neurotoxicity appears to respond to aggressive supportive care and to withholding chemotherapy [118–121]. Uridine triacetate should be considered if the patient otherwise meets established criteria.

Treatment for hand-and-foot syndrome is similarly supportive (pain control and wound care as needed). Immersing hands and feet in cold water and applying topical emollients may be beneficial [101]. It is also generally recommended that treatment be discontinued, or the dose be reduced, until symptoms resolve [11, 15, 100]. Dose adjustments should be made in consultation with the treating oncologist. Steroids are not recommended due to insufficient evidence of benefit [100].

Toxicity related to 5-FU's interaction with CYP2C9 should be managed supportively, and any long-term dose adjustments to medications such as warfarin or phenytoin should be made in consultation with the treating oncologist and/or the patient's prescribing physician [10, 77, 78, 126].

Granulocyte colony-stimulating factor (G-CSF) may be considered to treat myelosuppression, even in addition to uridine triacetate [26, 79, 90, 127, 128]. Recommendations on the administration of G-CSF are available from an article by Smith et al. on behalf of the American Society of Clinical Oncologic [128]. The decision to administer G-CSF should be made in conjunction with the treating oncologist.

Extracorporeal elimination has not been studied in the setting of overdose or early-onset severe toxicity. The FDA suggests that dialysis may reduce the plasma concentration of cytotoxic 5-FU metabolites [11]. Given these compounds' large volumes of distribution and rapid elimination in urine, and the lack of data to support this intervention, renal replacement therapy for fluoropyrimidines is only recommended for those patients who already require the therapy (e.g., end-stage renal disease, profound acidemia, etc.) [37, 69, 129].

Antidotal Therapy: Uridine Triacetate

Uridine triacetate (UTA) is the acetylated oral prodrug of uridine, a pyrimidine analogue consisting of uracil bound to a ribose ring. Once absorbed in the gut, it is deacetylated to form uridine. Uridine is subsequently converted to uridine triphosphate (UTP) which competes with FUTP for incorporation into RNA [80, 81, 126]. It has a time to peak serum concentration of 2–3 hours and is absorbed into cells via nonspecific nucleoside transporters [126]. It is metabolized via typical pyrimidine catabolic pathways and also excreted renally [126]. Uridine triacetate is dosed in adults as 10 grams every 6 hours for a total of 20 doses. In pediatric patients it is dosed at 6.2 grams/m²/dose, up to 10 grams, for a total of 20 doses [126]. It is typically well tolerated; adverse effects include vomiting, nausea, and diarrhea [80, 126].

According to the FDA, uridine triacetate is indicated in two settings. The first is after overdose of 5-FU or capecitabine, defined on the package insert as "the administration of fluorouracil at a dose, or infusion rate, greater than the intended dose or maximum tolerated dose for the patient's intended regimen of fluorouracil." Uridine triacetate is indicated in this setting regardless of the presence of symptoms. No distinction is made between intentional, unintentional, and iatrogenic overdoses. Its second indication is in the setting of "early-onset, severe or life-threatening toxicities affecting the cardiac or central nervous system, and/or early-onset, unusually severe adverse reactions (e.g., gastrointestinal toxicity and/or neutropenia)" that occur within 96 hours of the patient completing a 5-FU infusion or capecitabine administration [126].

The evidence for the use of uridine triacetate in humans comes largely from studies by Ison et al. in 2016 and Ma et al. in 2017, which compared the outcomes of patients in two open-label, single-arm trials to historical case controls. The studies demonstrate a difference in survival at 30 days of 96% in those who received uridine triacetate (97% in those with overdose and 86% in those with early-onset symptoms) versus 16% in those who did not. Of those patients who developed early-onset symptoms but started uridine triacetate over 96 hours after last dose administration, only 38% survived. All deaths among patients with early-onset symptoms occurred in this cohort. Of note, this study included both pediatric and adult patients, and several patients received UTA via nasogastric, orogastric, and gastrostomy tubes [80, 81]. Since these studies were published, multiple other authors have reported success in reversing toxicity with uridine triacetate [79, 130-132].

Ma et al. defined severe or life-threatening signs and symptoms as "severe cytotoxic mucosal and/or hematologic toxicities as well as acute encephalopathy and/or cardiomyopathy" [80]. The FDA's indications for uridine triacetate are broader. If possible, it is recommended to consult with the treating oncologist when deciding to administer UTA as this medication not only may undermine 5-FU's therapeutic effects but is expensive and potentially difficult to obtain [133, 134]. In situations where the degree of toxicity is unclear, or where it is unclear if toxicity requires therapy, it is recommended to consult with an oncologist and/or a toxicologist.

References

- Longley DB, Harkin DP, Johnston PG. 5-Fluorouracil: mechanisms of action and clinical strategies. Nat Rev Cancer. 2003;3(5):330–8. https://doi.org/10.1038/nrc1074.
- Rutman RJ, Cantarow A, Paschkis KE. Studies in 2-acetylaminofluorene carcinogenesis. III. The utilization of uracil-2-C14 by preneoplastic rat liver and rat hepatoma. Cancer Res. 1954;14(2):119–23.
- Diasio RB, Harris BE. Clinical pharmacology of 5-fluorouracil. Clin Pharmacokinet. 1989;16(4):215–37.
- Daher GC, Harris BE, Diasio RB. Metabolism of pyrimidine analogues and their nucleosides. Pharmacol Ther. 1990;48(2):189–222.
- National Cancer Institute Publications. NCI dictionaries. NCI dictionary of cancer terms. Bethesda: National Cancer Institute; 2020. https://www.cancer.gov/publications/dictionaries/cancer-terms.
- Lamont EB, Schilsky RL. The oral fluoropyrimidines in cancer chemotherapy. Clin Cancer Res. 1999;5(9):2289–96.
- Pu AT, Robertson JM. Current status of radiation sensitization by fluoropyrimidines. Oncology. 1995;9(8):707–14.
- Abele R, Alberto P, Kaplan S, Siegenthaler P, Hofmann V, Ryssel HJ, et al. Phase II study of doxifluridine in advanced colorectal adenocarcinoma. J Clin Oncol. 1983;1(12):750–4.
- Sakamoto J, Hamada C, Kodaira S, Nakazato H, Ohashi Y. Adjuvant therapy with oral fluoropyrimidines as main chemotherapeutic agents after curative resection for colorectal cancer: individual patient data meta-analysis of randomized trials. Jpn J Clin Oncol. 1999;29(2):78–86.
- Walko CM, Lindley C. Capecitabine: a review. Clin Ther. 2005;27(1):23–44. https://doi.org/10.1016/j.clinthera.2005.01.005.
- Xeloda (capecitabine) tablets, for oral use [package insert]. South San Francisco: Genentech; 1998.
- Fluorouracil injection, for intravenous use [package insert]. Irvine: Spectrum Pharmaceuticals Inc.; 1962.
- 13. National Institutes of Health, Public Health Service, HHS. Public teleconference regarding licensing and collaborative research opportunities for: methods and compositions relating to detecting dihydropyrimidine dehydrogenase (DPD). 3 July 2008. US National Archives. Federal Register. 73 FR 38233. https://www.federalregister.gov/documents/2008/07/03/E8-15182/public-teleconference-regarding-licensing-and-collaborative-research-opportunities-for-methods-and.
- Mikhail SE, Sun JF, Marshall JL. Safety of capecitabine: a review. Expert Opin Drug Saf. 2010;9(5):831–41.
- Cassidy J, Twelves C, Van Cutsem E, Hoff P, Bajetta E, Boyer M, et al. First-line oral capecitabine therapy in metastatic colorectal cancer: a favorable safety profile compared with intravenous 5-fluorouracil/leucovorin. Ann Oncol. 2002;13(4):566–75.
- 16. Meta-Analysis Group In Cancer, Lévy E, Piedbois P, Buyse M, Pignon JP, Rougier P, Ryan L, et al. Toxicity of fluorouracil in patients with advanced colorectal cancer: effect of administration schedule and prognostic factors. J Clin Oncol. 1998;16(11):3537–41.

- Lunenburg CA, Henricks LM, Guchelaar H-J, Swen JJ, Deenen MJ, Schellens JH, et al. Prospective DPYD genotyping to reduce the risk of fluoropyrimidine-induced severe toxicity: ready for prime time. Eur J Cancer. 2016;54:40–8.
- Amstutz U, Froehlich TK, Largiadèr CR. Dihydropyrimidine dehydrogenase gene as a major predictor of severe 5-fluorouracil toxicity. Pharmacogenomics. 2011;12(9):1321–36.
- Saltz LB, Cox JV, Blanke C, Rosen LS, Fehrenbacher L, Moore MJ, et al. Irinotecan plus fluorouracil and leucovorin for metastatic colorectal cancer. Irinotecan Study Group. N Engl J Med. 2000;343(13):905–14.
- 20. Rosmarin D, Palles C, Pagnamenta A, Kaur K, Pita G, Martin M, et al. A candidate gene study of capecitabine-related toxicity in colorectal cancer identifies new toxicity variants at DPYD and a putative role for ENOSF1 rather than TYMS. Gut. 2015;64(1):111–20.
- Milano G, Etienne MC, Pierrefite V, Barberi-Heyob M, Deporte-Fety R, Renee N. Dihydropyrimidine dehydrogenase deficiency and fluorouracil-related toxicity. Br J Cancer. 1999;79(3–4): 627–30.
- Ciccolini J, Gross E, Dahan L, Lacarelle B, Mercier C. Routine Dihydropyrimidine dehydrogenase testing for anticipating 5-fluorouracil–related severe toxicities: hype or hope? Clin Colorectal Cancer. 2010;9(4):224–8.
- Terrazzino S, Cargnin S, Re MD, Danesi R, Canonico PL, Genazzani AA. DPYD IVS14+1G>A and 2846A>T genotyping for the prediction of severe fluoropyrimidine-related toxicity: a metaanalysis. Pharmacogenomics. 2013;14(11):1255–72.
- 24. Meulendijks D, van Hasselt JGC, Huitema ADR, van Tinteren H, Deenen MJ, Beijnen JH, et al. Renal function, body surface area, and age are associated with risk of early-onset fluoropyrimidineassociated toxicity in patients treated with capecitabinebased anticancer regimens in daily clinical care. Eur J Cancer. 2016;54:120–30.
- 25. Schwab M, Zanger UM, Marx C, Schaeffeler E, Klein K, Dippon J, et al. Role of genetic and nongenetic factors for fluorouracil treatment-related severe toxicity: a prospective clinical trial by the German 5-FU Toxicity Study Group. J Clin Oncol. 2008;26(13):2131–8.
- 26. Tsalic M, Bar-Sela G, Beny A, Visel B, Haim N. Severe toxicity related to the 5-fluorouracil/leucovorin combination (the Mayo Clinic regimen): a prospective study in colorectal cancer patients. Am J Clin Oncol. 2003;26(1):103–6.
- Rothenberg ML, Meropol NJ, Poplin EA, Van Cutsem E, Wadler S. Mortality associated with irinotecan plus bolus fluorouracil/leucovorin: summary findings of an independent panel. J Clin Oncol. 2001;19(18):3801–7.
- Ma WW, Saif MW, El-Rayes BF, Fakih MG, Cartwright TH, Posey JA, et al. Emergency use of uridine triacetate for the prevention and treatment of life-threatening 5-fluorouracil and capecitabine toxicity. Cancer. 2017;123(2):345–56.
- Mercier C, Ciccolini J. Profiling dihydropyrimidine dehydrogenase deficiency in patients with cancer undergoing 5-fluorouracil/ capecitabine therapy. Clin Colorectal Cancer. 2006;6(4):288–96.
- 30. Loganayagam A, Arenas Hernandez M, Corrigan A, Fairbanks L, Lewis CM, Harper P, et al. Pharmacogenetic variants in the *DPYD*, *TYMS*, *CDA* and *MTHFR* genes are clinically significant predictors of fluoropyrimidine toxicity. Br J Cancer. 2013;108(12):2505–15.
- Dahan L, Ciccolini J, Evrard A, Mbatchi L, Tibbitts J, Ries P, et al. Sudden death related to toxicity in a patient on capecitabine and irinotecan plus bevacizumab intake: pharmacogenetic implications. J Clin Oncol. 2012;30(4):e41–e4.
- Wohlhueter RM, McIvor RS, Plagemann PGW. Facilitated transport of uracil and 5-fluorouracil, and permeation of orotic acid into cultured mammalian cells. J Cell Physiol. 1980;104(3):309–19.

- Schüller J, Cassidy J, Dumont E, Roos B, Durston S, Banken L, et al. Preferential activation of capecitabine in tumor following oral administration to colorectal cancer patients. Cancer Chemother Pharmacol. 2000;45(4):291–7.
- 34. Miwa M, Ura M, Nishida M, Sawada N, Ishikawa T, Mori K, et al. Design of a novel oral fluoropyrimidine carbamate, capecitabine, which generates 5-fluorouracil selectively in tumours by enzymes concentrated in human liver and cancer tissue. Eur J Cancer. 1998;34(8):1274–81.
- Budman DR, Meropol NJ, Reigner B, Creaven PJ, Lichtman SM, Berghorn E, et al. Preliminary studies of a novel oral fluoropyrimidine carbamate: capecitabine. J Clin Oncol. 1998;16(5):1795–802.
- 36. Ichikawa W, Takahashi T, Suto K, Sasaki Y, Hirayama R. Orotate phosphoribosyltransferase gene polymorphism predicts toxicity in patients treated with bolus 5-fluorouracil regimen. Clin Cancer Res. 2006;12(13):3928–34.
- Heggie GD, Sommadossi J-P, Cross DS, Huster WJ, Diasio RB. Clinical pharmacokinetics of 5-fluorouracil and its metabolites in plasma, urine, and bile. Cancer Res. 1987;47(8):2203–6.
- Danenberg PV, Danenberg KD. Effect of 5, 10-methylenetetrahydrofolate on the dissociation of 5-fluoro-2'-deoxyuridylate from thymidylate synthetase: evidence for an ordered mechanism. Biochemistry. 1978;17(19):4018–24.
- Santi DV, McHenry CS, Sommer H. Mechanism of interaction of thymidylate synthetase with 5-fluorodeoxyuridylate. Biochemistry. 1974;13(3):471–81.
- 40. Sommer H, Santi DV. Purification and amino acid analysis of an active site peptide from thymidylate synthetase containing covalently bound 5-fluoro-2'-deoxyuridylate and methylenetetrahydrofolate. Biochem Biophys Res Commun. 1974;57(3):689–95.
- Santi DV, McHenry CS. 5-Fluoro-2'-deoxyuridylate: covalent complex with thymidylate synthetase. Proc Natl Acad Sci U S A. 1972;69(7):1855–7.
- Mitrovski B, Pressacco J, Mandelbaum S, Erlichman C. Biochemical effects of folate-based inhibitors of thymidylate synthase in MGH-U1 cells. Cancer Chemother Pharmacol. 1994;35(2): 109–14.
- 43. Aherne GW, Hardcastle A, Raynaud F, Jackman AL. Immunoreactive dUMP and TTP pools as an index of thymidylate synthase inhibition; effect of tomudex (ZD1694) and a nonpolyglutamated quinazoline antifolate (CB30900) in L1210 mouse leukaemia cells. Biochem Pharmacol. 1996;51(10):1293–301.
- Ingraham HA, Tseng B, Goulian M. Nucleotide levels and incorporation of 5-fluorouracil and uracil into DNA of cells treated with 5-fluorodeoxyuridine. Mol Pharmacol. 1982;21(1):211–6.
- Cheng Y, Nakayama K. Effects of 5-fluoro-2'-deoxyuridine on DNA metabolism in HeLa cells. Mol Pharmacol. 1983;23(1):171–4.
- 46. Andersen S, Heine T, Sneve R, Konig I, Krokan HE, Epe B, et al. Incorporation of dUMP into DNA is a major source of spontaneous DNA damage, while excision of uracil is not required for cytotoxicity of fluoropyrimidines in mouse embryonic fibroblasts. Carcinogenesis. 2005;26(3):547–55.
- Vodenkova S, Buchler T, Cervena K, Veskrnova V, Vodicka P, Vymetalkova V. 5-Fluorouracil and other fluoropyrimidines in colorectal cancer: past, present and future. Pharmacol Ther. 2020;206:107447.
- Schuetz JD, Wallace HJ, Diasio RB. DNA repair following incorporation of 5-fluorouracil into DNA of mouse bone marrow cells. Cancer Chemother Pharmacol. 1988;21(3):208–10.
- 49. Tanaka M, Yoshida S, Saneyoshi M, Yamaguchi T. Utilization of 5-fluoro-2'-deoxyuridine triphosphate and 5-fluoro-2'deoxycytidine triphosphate in DNA synthesis by DNA polymerases α and β from calf thymus. Cancer Res. 1981;41(10):4132–5.
- Kufe D, Major P, Egan E, Loh E. 5-Fluoro-2'-deoxyuridine incorporation in L1210 DNA. J Biol Chem. 1981;256(17):8885–8.

- Major P, Egan E, Herrick D, Kufe D. 5-Fluorouracil incorporation in DNA of human breast carcinoma cells. Cancer Res. 1982;42(8):3005–9.
- 52. Houghton JA, Tillman DM, Harwood FG. Ratio of 2'-deoxyadenosine-5'-triphosphate/thymidine-5'-triphosphate influences the commitment of human colon carcinoma cells to thymineless death. Clin Cancer Res. 1995;1(7):723–30.
- 53. Yoshioka A, Tanaka S, Hiraoka O, Koyama Y, Hirota Y, Ayusawa D, et al. Deoxyribonucleoside triphosphate imbalance. 5-Fluorodeoxyuridine-induced DNA double strand breaks in mouse FM3A cells and the mechanism of cell death. J Biol Chem. 1987;262(17):8235–41.
- Lindahl T. An N-glycosidase from Escherichia coli that releases free uracil from DNA containing deaminated cytosine residues. Proc Natl Acad Sci U S A. 1974;71(9):3649–53.
- Grem JL, Fischer P. Enhancement of 5-fluorouracil's anticancer activity by dipyridamole. Pharmacol Ther. 1989;40(3):349–71.
- Kufe DW, Major PP. 5-Fluorouracil incorporation into human breast carcinoma RNA correlates with cytotoxicity. J Biol Chem. 1981;256(19):9802–5.
- Glazer RI, Lloyd LS. Association of cell lethality with incorporation of 5-fluorouracil and 5-fluorouridine into nuclear RNA in human colon carcinoma cells in culture. Mol Pharmacol. 1982;21(2):468–73.
- Carrico CK, Glazer RI. Effect of 5-fluorouracil on the synthesis and translation of polyadenylic acid-containing RNA from regenerating rat liver. Cancer Res. 1979;39(9):3694–701.
- Armstrong R, Lewis M, Stern S, Cadman E. Acute effect of 5-fluorouracil on cytoplasmic and nuclear dihydrofolate reductase messenger RNA metabolism. J Biol Chem. 1986;261(16): 7366–71.
- Dolnick B, Pink JJ. 5-fluorouracil modulation of dihydrofolate reductase RNA levels in methotrexate-resistant KB cells. J Biol Chem. 1983;258(21):13299–306.
- Ramberg ES, Ishaq M, Rulf S, Moeller B, Horowitz J. Inhibition of transfer RNA function by replacement of uridine and uridinederived nucleosides with 5-fluorouridine. Biochemistry. 1978;17(19):3978–85.
- Armstrong R, Takimoto C, Cadman E. Fluoropyrimidine-mediated changes in small nuclear RNA. J Biol Chem. 1986;261(1):21–4.
- Ghoshal K, Jacob ST. Specific inhibition of pre-ribosomal RNA processing in extracts from the lymphosarcoma cells treated with 5-fluorouracil. Cancer Res. 1994;54(3):632–6.
- Santi DV, Hardy LW. Catalytic mechanism and inhibition of trna (uracil-5-) methyltransferase: evidence for covalent catalysis. Biochemistry. 1987;26(26):8599–606.
- 65. Kanamaru R, Kakuta H, Sato T, Ishioka C, Wakui A. The inhibitory effects of 5-fluorouracil on the metabolism of preribosomal and ribosomal RNA in L-1210 cells in vitro. Cancer Chemother Pharmacol. 1986;17(1):43–6.
- 66. Randerath K, Tseng WC, Harris JS, Lu LJW. Specific effects of 5-fluoropyrimidines and 5-azapyrimidines on modification of the 5 position of pyrimidines, in particular the synthesis of 5-methyluracil and 5-methylcytosine in nucleic acids. In: Nass G, editor. Modified nucleosides and cancer. Recent results in cancer research/Fortschritte der Krebsforschung/Progrès dans les recherches sur Ie cancer, vol. 84. Berlin, Heidelberg: Springer; 1983. https://doi.org/10.1007/978-3-642-81947-6_22.
- Patton JR. Ribonucleoprotein particle assembly and modification of U2 small nuclear RNA containing 5-fluorouridine. Biochemistry. 1993;32(34):8939–44.
- Doong S-L, Dolnick B. 5-fluorouracil substitution alters pre-mRNA splicing in vitro. J Biol Chem. 1988;263(9):4467–73.
- Diasio RB, Beavers T, Carpenter J. Familial deficiency of dihydropyrimidine dehydrogenase. Biochemical basis for familial pyrimi-

dinemia and severe 5-fluorouracil-induced toxicity. J Clin Invest. 1988;81(1):47–51.

- Casale F, Canaparo R, Serpe L, Muntoni E, Pepa CD, Costa M, et al. Plasma concentrations of 5-fluorouracil and its metabolites in colon cancer patients. Pharmacol Res. 2004;50(2):173–9.
- Fraile RJ, Baker LH, Buroker TR, Horwitz J, Vaitkevicius V. Pharmacokinetics of 5-fluorouracil administered orally, by rapid intravenous and by slow infusion. Cancer Res. 1980;40(7):2223–8.
- 72. Fleming GF, Schilsky RL, Schumm LP, Meyerson A, Hong AM, Vogelzang NJ, et al. Phase I and pharmacokinetic study of 24-hour infusion 5-fluorouracil and leucovorin in patients with organ dysfunction. Ann Oncol. 2003;14(7):1142–7.
- Chirstophidis N, Vajda F, Lucas I, Drummer O, Moon W, Louis W. Fluorouracil therapy in patients with carcinoma of the large bowel: a pharmacokinetic comparison of various rates and routes of administration. Clin Pharmacokinet. 1978;3(4):330–6.
- Douglass H Jr, Mittelman A. Metabolic studies of 5-fluorouracil— II. Influences of the route of administration on the dynamics of distribution in man. Cancer. 1974;34(6):1878–81.
- 75. Twelves C, Glynne-Jones R, Cassidy J, Schuller J, Goggin T, Roos B, et al. Effect of hepatic dysfunction due to liver metastases on the pharmacokinetics of capecitabine and its metabolites. Clin Cancer Res. 1999;5(7):1696–702.
- Reigner B, Blesch K, Weidekamm E. Clinical pharmacokinetics of capecitabine. Clin Pharmacokinet. 2001;40(2):85–104.
- Kolesar JM, Johnson CL, Freeberg BL, Berlin JD, Schiller JH. Warfarin-5-FU interaction--a consecutive case series. Pharmacotherapy. 1999;19(12):1445–9.
- Brickell K, Porter D, Thompson P. Phenytoin toxicity due to fluoropyrimidines (5FU/capecitabine): three case reports. Br J Cancer. 2003;89(4):615–6.
- Grogan NSJ, Gaspar K, Lyckholm L. Management of 5-fluorouracil overdose with uridine triacetate in a patient unable to tolerate oral administration of antidote. Austin J Med Oncol. 2017;4(1):1031.
- Ma WW, Cartwright TH, El-Rayes BF, Posey J, Fakih M, King T, et al. Uridine triacetate as a life-saving antidote to capecitabine toxicity. J Clin Oncol. 2016;34(15 Suppl):e21612.
- 81. Ison G, Beaver JA, McGuinn WD, Palmby TR, Dinin J, Charlab R, et al. FDA approval: uridine triacetate for the treatment of patients following fluorouracil or capecitabine overdose or exhibiting early-onset severe toxicities following administration of these drugs. Clin Cancer Res. 2016;22(18):4545–9.
- 82. Mercier C, Dupuis C, Blesius A, Fanciullino R, Yang CG, Padovani L, et al. Early severe toxicities after capecitabine intake: possible implication of a cytidine deaminase extensive metabolizer profile. Cancer Chemother Pharmacol. 2009;63(6):1177–80.
- Serdjebi C, Milano G, Ciccolini J. Role of cytidine deaminase in toxicity and efficacy of nucleosidic analogs. Expert Opin Drug Metab Toxicol. 2015;11(5):665–72.
- 84. Johnson MR, Hageboutros A, Wang K, High L, Smith JB, Diasio RB. Life-threatening toxicity in a dihydropyrimidine dehydrogenase-deficient patient after treatment with topical 5-fluorouracil. Clin Cancer Res. 1999;5(8):2006–11.
- 85. Jennings BA, Kwok CS, Willis G, Matthews V, Wawruch P, Loke YK. Functional polymorphisms of folate metabolism and response to chemotherapy for colorectal cancer, a systematic review and meta-analysis. Pharmacogenet Genomics. 2012;22(4):290–304.
- Pullarkat S, Stoehlmacher J, Ghaderi V, Xiong Y, Ingles S, Sherrod A, et al. Thymidylate synthase gene polymorphism determines response and toxicity of 5-FU chemotherapy. Pharmacogenomics J. 2001;1(1):65–70.
- 87. Saif MW, Syrigos K, Mehra R, Mattison LK, Diasio RB. Dihydropyrimidine dehydrogenase deficiency (DPD) in GI malignancies: experience of 4-years. Pakistan J Med Sci Q. 2007;23(6):832.

- Morel A, Boisdron-Celle M, Fey L, Soulie P, Craipeau MC, Traore S, et al. Clinical relevance of different dihydropyrimidine dehydrogenase gene single nucleotide polymorphisms on 5-fluorouracil tolerance. Mol Cancer Ther. 2006;5(11):2895–904.
- Zhang H, Li Y-M, Zhang H, Jin X. DPYD* 5 gene mutation contributes to the reduced DPYD enzyme activity and chemotherapeutic toxicity of 5-FU. Med Oncol. 2007;24(2):251–8.
- 90. Raida M, Schwabe W, Häusler P, Van Kuilenburg AB, Van Gennip AH, Behnke D, et al. Prevalence of a common point mutation in the dihydropyrimidine dehydrogenase (DPD) gene within the 5'-splice donor site of intron 14 in patients with severe 5-fluorouracil (5-FU)-related toxicity compared with controls. Clin Cancer Res. 2001;7(9):2832–9.
- 91. Van Kuilenburg AB, Meinsma R, Zoetekouw L, Van Gennip AH. High prevalence of the IVS14+ 1G> A mutation in the dihydropyrimidine dehydrogenase gene of patients with severe 5-fluorouracil-associated toxicity. Pharmacogenet Genomics. 2002;12(7):555–8.
- 92. van Kuilenburg AB, Haasjes J, Richel DJ, Zoetekouw L, Van Lenthe H, De Abreu RA, et al. Clinical implications of dihydropyrimidine dehydrogenase (DPD) deficiency in patients with severe 5-fluorouracil-associated toxicity: identification of new mutations in the DPD gene. Clin Cancer Res. 2000;6(12):4705–12.
- Yen JL, McLeod HL. Should DPD analysis be required prior to prescribing fluoropyrimidines? Eur J Cancer. 2007;43(6): 1011–6.
- 94. Etienne MC, Lagrange JL, Dassonville O, Fleming R, Thyss A, Renee N, et al. Population study of dihydropyrimidine dehydrogenase in cancer patients. J Clin Oncol. 1994;12(11): 2248–53.
- 95. Lu Z, Zhang R, Diasio RB. Dihydropyrimidine dehydrogenase activity in human peripheral blood mononuclear cells and liver: population characteristics, newly identified deficient patients, and clinical implication in 5-fluorouracil chemotherapy. Cancer Res. 1993;53(22):5433–8.
- Burns R, Beland SS. Effect of biological time on the determination of the LD50 of 5-fluorouracil in mice. Pharmacology. 1984;28(5):296–300.
- Johnson RK, Garibjanian BT, Houchens DP, Kline I, Gaston MR, Syrkin AB, et al. Comparison of 5-fluorouracil and ftorafur. I. Quantitative and qualitative differences in toxicity to mice. Cancer Treat Rep. 1976;60(9):1335–45.
- Liu J, Skradis A, Kolar C, Kolath J, Anderson J, Lawson T, et al. Increased cytotoxicity and decreased in vivo toxicity of FdUMP [10] relative to 5-FU. Nucleosides Nucleotides. 1999;18(8):1789–802.
- 99. Milano G, Etienne-Grimaldi MC, Mari M, Lassalle S, Formento JL, Francoual M, et al. Candidate mechanisms for capecitabine-related hand–foot syndrome. Br J Clin Pharmacol. 2008;66(1):88–95.
- Gressett SM, Stanford BL, Hardwicke F. Management of handfoot syndrome induced by capecitabine. J Oncol Pharm Pract. 2006;12(3):131–41.
- 101. Lassere Y, Hoff P. Management of hand-foot syndrome in patients treated with capecitabine (Xeloda®). Eur J Oncol Nurs. 2004;8:S31–40.
- Labianca R, Beretta G, Clerici M, Fraschini P, Luporini G. Cardiac toxicity of 5-fluorouracil: a study on 1083 patients. Tumori J. 1982;68(6):505–10.
- Jensen SA, Sørensen JB. Risk factors and prevention of cardiotoxicity induced by 5-fluorouracil or capecitabine. Cancer Chemother Pharmacol. 2006;58(4):487–93.
- Akhtar SS, Salim KP, Bano ZA. Symptomatic cardiotoxicity with high-dose 5-fluorouracil infusion: a prospective study. Oncology. 1993;50(6):441–4.

- 105. Koca D, Salman T, Unek IT, Oztop I, Ellidokuz H, Eren M, et al. Clinical and electrocardiography changes in patients treated with capecitabine. Chemotherapy. 2011;57(5):381–7.
- 106. Saif MW, Shah MM, Shah AR. Fluoropyrimidine-associated cardiotoxicity: revisited. Expert Opin Drug Saf. 2009;8(2):191–202.
- 107. Kelly C, Bhuva N, Harrison M, Buckley A, Saunders M. Use of raltitrexed as an alternative to 5-fluorouracil and capecitabine in cancer patients with cardiac history. Eur J Cancer. 2013;49(10):2303–10.
- 108. Sorrentino MF. 5-fluorouracil induced cardiotoxicity: review of the literature. Cardiol J. 2012;19:453–8.
- 109. Rezkalla S, Kloner RA, Ensley J, Al-Sarraf M, Revels S, Olivenstein A, et al. Continuous ambulatory ECG monitoring during fluorouracil therapy: a prospective study. J Clin Oncol. 1989;7(4):509–14.
- Anand AJ. Fluorouracil cardiotoxicity. Ann Pharmacother. 1994;28(3):374–8.
- 111. Polk A, Vistisen K, Vaage-Nilsen M, Nielsen DL. A systematic review of the pathophysiology of 5-fluorouracil-induced cardiotoxicity. BMC Pharmacol Toxicol. 2014;15(1):47.
- 112. Robben NC, Pippas AW, Moore JO. The syndrome of 5-fluorouracil cardiotoxicity. An elusive cardiopathy. Cancer. 1993;71(2):493–509.
- 113. Fontanella C, Aita M, Cinausero M, Aprile G, Baldin MG, Dusi V, et al. Capecitabine-induced cardiotoxicity: more evidence or clinical approaches to protect the patients' heart? Onco Targets Ther. 2014;7:1783–01.
- 114. Kosmas C, Kallistratos MS, Kopterides P, Syrios J, Skopelitis H, Mylonakis N, et al. Cardiotoxicity of fluoropyrimidines in different schedules of administration: a prospective study. J Cancer Res Clin Oncol. 2008;134(1):75–82.
- 115. Trotti A, Byhardt R, Stetz J, Gwede C, Corn B, Fu K, et al. Common toxicity criteria: version 2.0. An improved reference for grading the acute effects of cancer treatment: impact on radiotherapy. Int J Radiat Oncol Biol Phys. 2000;47(1):13–47.
- 116. Ducreux M, Bennouna J, Hebbar M, Ychou M, Lledo G, Conroy T, et al. Capecitabine plus oxaliplatin (XELOX) versus 5-fluorouracil/leucovorin plus oxaliplatin (FOLFOX-6) as first-line treatment for metastatic colorectal cancer. Int J Cancer. 2011;128(3):682–90.
- 117. Mercier C, Ciccolini J. Severe or lethal toxicities upon capecitabine intake: is DPYD genetic polymorphism the ideal culprit? Trends Pharmacol Sci. 2007;28(12):597–8.
- Moore DH, Fowler WC Jr, Crumpler LS. 5-Fluorouracil neurotoxicity. Gynecol Oncol. 1990;36(1):152–4.
- 119. Cordier P-Y, Nau A, Ciccolini J, Oliver M, Mercier C, Lacarelle B, et al. 5-FU-induced neurotoxicity in cancer patients with profound DPD deficiency syndrome: a report of two cases. Cancer Chemother Pharmacol. 2011;68(3):823–6.
- 120. Formica V, Leary A, Cunningham D, Chua YJ. 5-Fluorouracil can cross brain-blood barrier and cause encephalopathy: should we expect the same from capecitabine? A case report on capecitabineinduced central neurotoxicity progressing to coma. Cancer Chemother Pharmacol. 2006;58(2):276–8.

- 121. Couch LS, Groteluschen DL, Stewart JA, Mulkerin DL. Capecitabine-related neurotoxicity presenting as trismus. Clin Colorectal Cancer. 2003;3(2):121–3.
- Abbott JD, Curtis JP, Murad K, Kramer HM, Remetz MS, Setaro JF, et al. Spontaneous coronary artery dissection in a woman receiving 5-fluorouracil: a case report. Angiology. 2003;54(6):721–4.
- 123. De Forni M, Malet-Martino MC, Jaillais P, Shubinski R, Bachaud J, Lemaire L, et al. Cardiotoxicity of high-dose continuous infusion fluorouracil: a prospective clinical study. J Clin Oncol. 1992;10(11):1795–801.
- Dalzell JR, Samuel LM. The spectrum of 5-fluorouracil cardiotoxicity. Anti-Cancer Drugs. 2009;20(1):79–80.
- Eskilsson J, Albertsson M. Failure of preventing 5-fluorouracil cardiotoxicity by prophylactic treatment with verapamil. Acta Oncol. 1990;29(8):1001–3.
- 126. Vistoguard (uridine triacetate) oral granules [package insert]. West Conshohocken: Wellstat Therapeutics Corporation; 2015.
- 127. Andreica IW, Pfeifer E, Rozov M, Tavares E, Shakurova A, Ortiz T. Fluorouracil overdose: clinical manifestations and comprehensive management during and after hospitalization. J Hematol Oncol Pharm. 2015;5(2):43–7.
- 128. Smith TJ, Bohlke K, Lyman GH, Carson KR, Crawford J, Cross SJ, et al. Recommendations for the use of WBC growth factors: American Society of Clinical Oncology clinical practice guideline update. J Clin Oncol. 2015;33(28):3199–212.
- 129. Ezzeldin H, Diasio R. Dihydropyrimidine dehydrogenase deficiency, a pharmacogenetic syndrome associated with potentially life-threatening toxicity following 5-fluorouracil administration. Clin Colorectal Cancer. 2004;4(3):181–9.
- Saif MW, Diasio RB. Benefit of uridine triacetate (Vistogard) in rescuing severe 5-fluorouracil toxicity in patients with dihydropyrimidine dehydrogenase (DPYD) deficiency. Cancer Chemother Pharmacol. 2016;78(1):151–6.
- 131. Santos C, Morgan BW, Geller RJ. The successful treatment of 5-fluorouracil (5-FU) overdose in a patient with malignancy and HIV/AIDS with uridine triacetate. Am J Emerg Med. 2017;35(5):802 e7–e8.
- 132. Vaudo CE, Gil B, Galuski K, Zarwan C, Nugent FW. Early-onset 5-fluorouracil toxicity in a patient negative for dihydropyrimidine dehydrogenase mutations: the clinical course of reversal with uridine triacetate. Pharmacotherapy. 2016;36(11):e178–e82.
- 133. Lo LL, Lingaratnam S, Rowe C, Phillips KA. 239 An Australian experience of the use of uridine triacetate (Vistogard) antidote for severe capecitabine toxicity–a case report. In: Proceedings of COSA's 43rd and ANZBCTG's 38th Annual Scientific Meetings: Partners for Progress in Breast Cancer Research and Care, Queensland, Australia, Nov 2016. Asia Pac J Clin Oncol. 2016;12(Suppl 5):239. https://onlinelibrary.wiley.com/ toc/17437563/2016/12/S5.
- 134. Dart RC, Stark Y, Fulton B, Koziol-McLain J, Lowenstein SR. Insufficient stocking of poisoning antidotes in hospital pharmacies. JAMA. 1996;276(18):1508–10.

Part V Palliative Care

Emergency Department Use at End of Life

Hsien Seow and Kayla McMillan

Introduction

As cancer treatments advance and patients with cancer are living longer, there is also a growing potential for increased visits to the emergency department (ED) at end of life. Endof-life (EOL) care specifically refers to the final weeks of a patient's life, when symptoms commonly increase in intensity and death approaches [1]. EOL care is related to palliative care, which is a holistic approach to care, with emphasis on preventing or relieving suffering and improving quality of life [2]. Stakeholders have been advocating for improved EOL care and earlier integration of palliative care into the cancer trajectory for over two decades (Fig. 55.1) [3]. The American Society of Clinical Oncologic endorsed a clinical practice guideline supporting the provision of early palliative care concurrently with standard oncologic care in 2017 [4]. Without effective EOL or palliative care, many dying cancer patients have unmet clinical needs, uncontrolled symptoms, and poor quality of life, as well as fear, anxiety, and depression, which may cause them to visit the ED [5]. Avoiding unnecessary ED visits may lead to economic benefits (ED visits and the ensuing hospital care are costly), improve patient experience (many patients do not want to be cared for in a hospital or ED at the EOL), and improve quality of care (some causes of ED visits are avoidable with proper planning).

The ED is a setting of care focused on the management of acutely presenting medical problems. It is a fast-paced work environment with an emphasis on identifying the problem and instituting a solution over a very short interval of time. In contrast, dying cancer patients typically have complex medi-

H. Seow (⊠)

Department of Oncology, McMaster University, Hamilton, ON, Canada e-mail: seowh@mcmaster.ca

K. McMillan Department of Oncology, Juravinski Cancer Centre, Hamilton, ON, Canada



cal histories, multiple symptoms, and difficult psychosocial circumstances. Therefore, optimal EOL care requires a great deal of time to honestly discuss prognostic information, make clear recommendations, facilitate patient-family discussions, affirm patient choices, address holistic/psychosocial aspects of well-being, and plan for unexpected changes in a patient's condition. This is the antithesis of care typically provided in the ED. EOL trajectories in the ED have been characterized as being either "spectacular" (e.g., sudden, traumatic death) or "subtacular" (e.g., slow process of dying or a nonemergency death) [6]. ED staff are well trained to deal with "spectacular" deaths but often distance themselves from patients dying "subtacular" deaths. As a result, EOL



Late initiation of palliative care

[©] Springer Nature Switzerland AG 2021 K. H. Todd et al. (eds.), *Oncologic Emergency Medicine*, https://doi.org/10.1007/978-3-030-67123-5_55

care in ED is typically far from ideal. Hence, ED visits at the EOL are a widely used indicator of poor-quality care [7, 8].

There have been a few literature reviews on EOL care in the ED that were not cancer-specific. One review focused on the evidence for managing dying patients in the ED and identifying areas for improvement [9]. The authors observed that of the 160 papers included in the overview, the 6 main themes that arose were (1) the uncertainty of treatment in ED for patients at EOL; (2) quality of life issues; (3) costs; (4) ethical and social issues; (5) interaction between ED and other services; and (6) strategies for out-of-hospital care. Another similar review focused on barriers to providing optimal care in the ED. Of the 27 studies included, they identified minimal resources, lack of staff education, and outdated models of care as the main challenges [10]. A third review focused on older adults at EOL in the ED and identified 14 articles [11]. The review noted there was limited evidence regarding the definition, clinical profile, care delivery, and outcomes for older people requiring EOL care in the ED, though they did note that there were multiple tools available for dealing with EOL care, including predicting mortality and assessing functional status, comorbidities, symptom distress, palliative care needs, quality of life, and caregiver's stress. Nonetheless, the growing numbers of older adults, the increased pressures of access to high-quality care in a timely manner, and the escalating costs of healthcare delivery are bringing more attention to appropriate use of ED services, especially for cancer patients at the EOL with well-documented symptom needs.

This chapter will review the topic of ED use at the EOL with a specific focus on cancer patients. It will review:

- 1. The rationale for investigating ED use at EOL
- 2. The frequency with which ED visits occur
- 3. Common reasons for ED visits at the EOL
- 4. Factors associated with ED visits at the EOL
- 5. Evidence for strategies to mitigate ED use at the EOL

Case Study

Mr. X, a 75-year-old white male, presents to the ED at 11:00 pm with increasing abdominal pain and worsening fatigue over the past several days. He is accompanied with his wife, who is his primary caregiver at home. Mr. X has been a patient of your hospital for many years, and as such you look up his records in the hospital EMR. From the EMR you learn that Mr. X was diagnosed with squamous cell carcinoma of the lung approximately 7 years ago and underwent a wedge resection 5 years ago. Since his lung resection, he has also had chemotherapy. He and his wife live in a sub-urb outside of town, which is where his primary care doctor is located. Also from the EMR, you learn that approximately

5 months ago, Mr. X. was diagnosed with a metastasis in his brain. He has been followed by his oncologist in the outpatient setting. You can also see that he has not met with the palliative care service which is available at your hospital.

After you have read about his history in the EMR report, you decide to ask Mr. X and his wife some questions to gain his understanding about his illness and his current symptoms. After discussing his abdominal pain, you rule out any other contributing factors, and provide symptom management for him, as well as you provide a prescription to last him until he is scheduled to see his family doctor in the community next.

After assessing for any other contributing factors, it is likely that his fatigue is related to his decline in overall functioning. After speaking with him and his wife, you understand that he is unable to get out of bed most days, and he is having trouble walking without assistance. You ask Mr. X about his outpatient follow-up, in order to ascertain whether he has been seen by a palliative care specialist in the community. His wife lets you know that he has not; no one has suggested this to them. You take a few minutes to explain why a palliative care service is appropriate for him, and after letting him discuss it with his wife, they agree to be referred to the service at your hospital so they can see someone quickly. You refer him to your inhospital palliative care team, and request an urgent consult, given his advanced illness and his steady decline. Once the palliative care team member arrives, you give handover to ensure seamless transition and that no details are missed, including letting them know you gave him a prescription for pain medicine.

Rationale for Investigating ED Use at EOL

Indicator of Poor-Quality Care

In 2003, Earle et al. published a landmark paper describing quality indicators of EOL cancer care that could be measured with administrative healthcare data [7]. The impact of this publication was significant because (1) stakeholders began to discuss the merits of the quality indicators, selected through literature review, focus groups with patients and family members, and an expert panel, and (2) researchers began measuring them using readily available administrative healthcare data. One of the key indicators of poor-quality care identified was frequent ED visits near the EOL. It became operationalized as more than one ED visit in the last 30 days of life [8]. The rationale for this indicator is because high rates of unplanned medical encounters at EOL, exemplified by ED visits, may indicate overuse of aggressive care, inattention to symptom issues, poor planning by providers to anticipate patient needs, insufficient support or education for the caregivers, lack of advance directives, or inadequate availability of palliative care resources, such as hospice services; when these elements are provided or addressed, they can help avoid unnecessary ED visits at EOL. Examples of other poor-quality care measures that emerged from this publication include a short interval of time between chemotherapy and death, ICU admissions near EOL, and a short time interval between hospice enrollment and death.

Patients and Families' Perspective on ED Visits at EOL

Cancer patients will frequently visit the ED when symptoms are uncontrolled and they find themselves with unmet needs [12]. Many would prefer care by their usual provider or someone from the cancer center where they typically receive care. However, when the need arises and patients present at their local ED, it presents a challenge to the emergency physician [13].

Most reports indicate that patients prefer to be at home at the EOL [14–17]. Earle et al.'s original indicator [7] was developed by including perspectives from 12 patients with incurable cancer and 4 family members: they associated ED visits primarily with toxicity management of aggressive chemotherapy. Early in the trajectory, this was expected and described as a necessary part of receiving therapy. Later in the trajectory of care, patients' willingness to tolerate toxicity for non-curative therapy diminished. In turn, their willingness to endure ED visits or accept them as a necessary component of care also diminished. Interestingly, in a study of Japanese bereaved family members, only 14% endorsed the appropriateness of frequency of ED visits as an indicator of poor-quality care. This suggests that this particular measure may be perceived differently in different cultures where the perspective on a "good death" may be different [18].

In a study of stakeholder perspective of quality indicators for EOL care, 16 women with metastatic breast cancer and 8 bereaved family caregivers participated in focus groups to provide their perspectives on quality indicators at EOL [19]. The dominant themes that emerged were support for, access to, and early enrollment for palliative care services, continuity of care, and multidisciplinary care. The authors do not describe a dominant theme related to ED visits, though ED visits clearly disrupt continuity of care in the ambulatory setting and lack of adequate access to palliative care may lead to ED visits.

Healthcare Provider Perspective on ED Visits at EOL

Healthcare providers strongly endorsed ED visits as a quality indicator. In a Delphi process conducted in Canada with healthcare professionals evaluating the acceptability of fre-

quency of ED visits near the EOL, more than 80% of participants agreed the indicator was meaningful and important [19]. An American qualitative study of perspectives of care providers in the ED, including physicians, nurses, and other providers, revealed several emerging themes. These include conflict among providers about the feasibility and desirability of providing palliative care in the ED, not seeing a difference between palliative care and EOL care, poor communication between providers in the inpatient and outpatient setting, conflicts about withholding life-prolonging treatment, and inadequate training in pain management. A similar study from Germany describes ED physicians as having difficulty dealing with palliative patients because of uncertainty regarding aspects of psychosocial care and EOL decision-making [20]. Relatedly, a survey from Australia indicates that ED physicians feel that the care they provide to EOL cancer patients is futile [21]. These sentiments conceptually endorse the notion that, from a provider perspective, the ED may not be the best place to care for this patient population.

System Perspective on ED Visits at the EOL

In some jurisdictions, ED use at EOL is being measured at a population level and has even become a system-wide metric in some countries. In Australia, routine data from the Department of Veterans Affairs showed that cancer decedents averaged one ED visit in the last 6 months of life [22]. Another study in Australia examined the ED use in the last year of life among cancer and non-cancer patients revealing that 70% had at least one ED visit [23].

In the UK, the National EOL Care Intelligence Network, now a part of Public Health England, is measuring ED use for quality improvement. In 2018, a study by Henson et al. discovered that as many as 30.7% of patients in the UK who died of any cancer had at least one ED visit [24]. In another UK study for urological cancer patients in the last year of life, they found that emergency admissions tended to be significantly longer and more costly than planned admissions to hospital. As well, emergency admissions were far more common than planned admissions to hospital in the three largest urological cancer groups [25].

In the USA, the Veterans Health Affairs Department has reported data on ED visits at end of life [26]. At a national level, the American Society for Clinical Oncologic has included the "percentage of patients who died from cancer with more than one emergency department visit in the last 30 days of life" as a measure for voluntary reporting for their end-of-life module in the Quality Oncologic Practice Initiatives (QOPI) [27]. For instance, one cancer center reported their ED visit admission rate as 68% in their last month of life, compared to peer healthcare institutions reporting an average of 32% [28]. Nonetheless, the difficulty of risk stratification to separate those for whom the ED visit was appropriate versus avoidable continues to be a challenge for its use as a national quality indicator for accreditation [29].

In Canada, the cancer agency for Ontario has created the "Cancer System Quality Index" which for many years included ED visits in the last 14 days of life as a quality indicator (www.csqi.on.ca). The index, published annually on the Internet, includes a plethora of quality indicators for cancer across the care trajectory. For instance, in their 2016 report, they noted that 40% of cancer patients visited the ED in the last 2 weeks of life [30]. Nationally, the Canadian Partnership Against Cancer has begun measuring this quality indicator in their 2017 report, they noted that 49% of cancer patients who died in hospital visited the ED once, 25% twice or more, and 27% no times in their final 28 days of life [31].

Frequency of ED Visits at EOL

The frequency of ED visits depends on several definitional and contextual factors. Table 55.1 [7, 23, 31–51] lists a review of publications describing the frequency of ED visits at the EOL. These publications typically used populationbased retrospective cohort studies linking large administrative data. They were conducted in Canada, the USA, Australia, Egypt, England, Mexico, and Taiwan, ranging in cohort sizes from 154 to 272,832, where years of study periods spanned 1991–2018. The prevalence of ED visits ranged from 1.5% having two or more ED visits in the last 30 days of life [32] to 85% having an ED visit in the last 6 months of life [33]. The duration of the observation window from death varies from 2 weeks to 1 year, with 30 days being most common. The patients in the numerator may be counted with any ED visits, but in several papers, more than one visit is required. Many studies are population based, but some are institutional. Some include all cancer types, where others include only specific cancers. These differences in study design, definitions, and inclusion/exclusion criteria should be considered when making comparisons.

The majority of studies used an observation window looking backwards from the date of death to create a decedent cohort (i.e., people who have died) rather than a prospective cohort that followed patients until they all died. This approach has been criticized since people who have died may not be the same people one would identify as actively dying [53]. However, when used with administrative data, this approach allows for easy identification of relevant cohorts, efficient study of all patients who died (rather than a nonrandom subset), and timely evaluations for quality improvement [54]. Contextual factors must also be considered. Specifically, at are the palliative care supports available for study par-

H. Seow and K. McMillan

what are the palliative care supports available for study participants, and more generally, how is care structured? For example, in Warren et al., ED use was compared in Ontario, Canada, and the National Cancer Institute Surveillance, Epidemiology, and End Results Program (SEER) regions in the USA. The methods of the study explicitly harmonized all definitions in both jurisdictions to ensure a fair comparison. ED rates differed by about 10%, which is likely a reflection of differences in health policies, structures, and process of care in the two settings, such as access to home care, medications, and hospice services [43].

Despite these differences, most authors conclude that ED use in cancer patients at EOL is too high and that this population-based metric should be driven down as low as possible. Studies examining regional variation have attempted to define a benchmark rate. Earle et al. in 2005 described the regional variation in ED visit rates at the EOL among Medicare beneficiaries with cancer; they found almost threefold variation among 11 different regions in the USA [55]. They defined an empiric benchmark as the top decile of performers to derive a benchmark of <4% with more than one ED visit in the last 30 days. This benchmark was achieved in only one other study in a community setting [56]. In fact, the majority of the studies with a similar definition exceed this proposed benchmark. For instance, Barbera et al. examined 33 health regions in 3 provinces in Canada, and the top decile had a 34% ED visit rate [57]. While negative publication bias may partially explain this, the existing data support the impression that improvements in care are required. It is important to remember that this measure is meant to be considered at a population level. It would be impossible to have a system where the value of this measure is zero. ED care may be entirely appropriate for a particular individual. But when population values are in excess of 30%, one ought to further examine the health policies and systems of care. This was reinforced by a Canadian study of cancer decedents by Seow et al. which showed ED visit rates at EOL varied more greatly by provinces than by community population size [58].

Reasons for ED Visits at the EOL

There have been several studies examining why cancer patients go to the ED, though most do not restrict their evaluation to the EOL period. Mayer et al. conducted a study identifying over four million ED visits in North Carolina, USA [59]. The top three complaints were related to pain, respiratory distress, and gastrointestinal issues. Beyond treatment, in a Korean study of over 5500 cancer-specific ED visits, 55% of visits were related in some way to disease progression [60]. This disease progression often leads to worsening

Table 55.1	Frequency	of ED	visits	at end	of life
------------	-----------	-------	--------	--------	---------

	Year of					Time window	
First author	publication	Country	Jurisdiction	Cancer type	Cohort size	prior to death	Patients to ED (%)
Allende-Pérez et al. [34]	2020	Mexico	Mexico City	Any cancer	426	30 d	8.9% >= 3 ED admissions
Henson et al. [24]	2018	England	England	Any cancer	124,030	30 d	30.7% 1 visit; 5.1% >=1 visit
Falchook et al. [32]	2017	USA	Five geographic regions	Lung, CRC, breast, GI, prostate	28,731	30 d	1.5–2.5% for >=2 ED visits
Seow et al. [33]	2016	Canada	Ontario	Any cancer	54,576	6 mo	85%
Alsirafy et al. [35]	2016	Egypt	Single institution	Terminal cancer	154	3 mo	77%
Obermeyer et al. [36]	2016	USA	SEER Medicare	Any cancers	272,832	6 mo	81%
Lee et al. [37]	2015	Taiwan	Taiwan	Any cancer	23,883	1 yr	81.5%
McNamara et al. [38]	2013	Australia	Western Australia	Any cancer	746	12 mo/90 d	65%/47%
Miesfeldt et al. [39]	2012	USA	SEER Medicare	Poor prognosis cancer	235,821	30 d	10% with >= 1 ED
Maddison et al. [40]	2012	Canada	Nova Scotia	CRC	1201	30 d	23.2%
Ho et al. [41]	2011	Canada	Ontario	All cancer deaths	227,161	30 d	8.6–10.5% with >= 1 ED visit
Saito et al. [42]	2011	USA	SEER Medicare	NSCLC	7879	30 d	23.2% with >= 1 ED visit
Warren et al. [43]	2011	USA/ Canada	SEER Medicare/ Ontario	NSCLC	13,533 US/8100 ON	30 d	37% US/49% ON
Rosenwax et al. [23]	2011	Australia	Western Australia	Cancer or nine other conditions	1071	12 mo	70%
Bergman et al. [44]	2011	USA	SEER Medicare	Prostate cancer	13,804	6 mo	mean # visits: 1.53
Barbera et al. [45]	2010	Canada	Ontario	Ovary, endometrium, cervix	2040	2 wk	34%
Keating et al. [46]	2010	USA	VA matched with SEER	Lung or CRC	2913 in each group	30 d	13.1% >=1 ED [VA]; 14.7% [SEER]
Setoguchi et al. [47]	2010	USA	Pennsylvania	Lung, breast, CRC, prostate	7565	30 d	38.90%
Barbera et al. [48]	2010	Canada	Ontario	Any cancer	91,561	6 mo/2 wk	83.8%/33.9% >=1 ED visit
Tang et al. [49].	2009	Taiwan	Taiwan	Any cancer	242,530	30 d	16-21%
Smith et al. [50]	2009	USA	SEER Medicare	Lung, CRC, breast, prostate	40,960	30 d	5-10% with >= 2 ED
Barbera et al. [51]	2008	Canada	Ontario	Lung cancer	5855	2 wk	32.20%
Earle et al. [8]	2004	USA	SEER Medicare	Lung, breast, CRC, other GI	28,777	30 d	7.2–9.2% with >=1 ED visit
Burge et al. [52]	2003	Canada	Nova Scotia	Any cancer	8702	After dx	mean # visits: 1.33

CRC colorectal cancer, GI gastrointestinal, SEsER Surveillance, Epidemiology, and End Results Program, NSCLC non-small cell lung cancer, VA US Department of Veterans Affairs

symptoms. Indeed, one study showed that physical symptom burden in the ambulatory cancer setting was strongly associated with the likelihood of an ED visit [61].

A systematic review of reasons for ED use by all cancer patients revealed that many visits are likely chemotherapyrelated toxicity as demonstrated by frequent visits for fever, neutropenia, and gastrointestinal complaints [62]. Another review also identified fever, urinary complaints, malnutrition, neutropenia, and gastrointestinal complaints as main reasons [63]. Similarly, a third study examined 18 EDs in the USA and found that three-quarters of cancer patients presenting to ED had undergone cancer treatment in the past 30 days and over 60% had advanced or metastatic cancer [64]. The five most common ED diagnoses were abdominal pain, fever, shortness of breath, nausea/vomiting, and throat/ chest pain [64].

A few studies focused on cancer ED visits at end of life. In one Canadian study of over 91,000 cancer patients, they identified 36,600 ED visits in the last 2 weeks of life. The main reasons for visits to ED in the last 2 weeks are listed in Table 55.2 Top ten reasons for ED visits in the last 2 weeks of

IXAIIK	Reasons	requency		
		(no.)	(%)	
1	Lung cancer	3242	8.9	
2	Dyspnea	1844	5.0	
3	Pneumonia	1832	5.0	
4	Abdominal pain	1126	3.1	
5	Malaise and fatigue	1084	3.0	
6	Palliative care	1042	2.9	
7	Dehydration	944	2.6	
8	Pleural effusion	717	2.0	
9	Altered consciousness	689	1.9	
10	Pancreatic cancer	585	1.6	

Table 55.2 [48]. Pain was responsible for about 5% of visits. As well, dyspnea, pneumonia, and pleural effusion were also very common. This study used administrative sources of data and was limited by the coding system for the diagnoses associated with each visit. As such, lung cancer was the most common "reason," although this does not actually reveal the cause of the visit. This likely reflects the impact of progression of disease in a common cancer type and is consistent with other works specific to lung cancer [51]. In a US study of cancer patients who died in the ED in the state of North Carolina, over one-third of those deaths had a cause of death due to lung cancer. Across all cancer deaths, the most common chief complaints were respiratory, gastrointestinal, and neurological [65]. Similarly, a Turkish study examining ED visits within a month of death identified dyspnea as the main complaint, with lung cancer as the most common cancer [66]. In the Netherlands, a study of advanced cancer patients visiting the ED within 3 months of death found that pain and dyspnea were the main reported symptoms [67]. A nationwide population-based study of Taiwanese cancer patients at end of life also identified "symptoms, signs, and ill defined conditions" as the main cause of admission for a third of cases, using ICD-9-CM diagnosis codes [37]. A study in Singapore looking at reasons for ED visits among cancer patients in the last week of life identified breathlessness as the cause for nearly 60% of patients, followed by pain (35%), general weakness or lethargy (27%), and decreased appetite (24%) [68].

Pain management warrants specific attention because it is commonly experienced by ambulatory cancer patients at the EOL [1] and often leads cancer patients to visit the ED [48, 59–62]. Two studies, one with Canadian patients and another from the USA, indicate that about a third of cancer patients have inadequately managed pain [69, 70]. In a study from Mexico, uncontrolled pain was also one of the principal reasons for ED visits [34]. This observation is essentially unchanged over the past two decades [71]. Meanwhile, emergency physicians have indicated they are not comfortable managing pain in this population [50]. Furthermore, overcrowding, a common ED problem, is associated with worse pain management [72]. Cancer patients deserve meticulous management of their pain by those best trained to do so.

When considering why patients visit the ED, a natural follow-up question is whether the visit was possibly avoidable or was the ED the best and only place for the patient to receive care? In a Canadian study, potentially avoidable visits were defined as ED visits related to a technical issue, such as catheter issues or prescription refills, and occurred about 1% of the time in the last 2 weeks of life. 8.4% of visits were for reasons such as malaise, fatigue, or need for palliative care, which likely represent some version of the patient "not coping" at home [48]. Arguably, these visits are also avoidable. In one small Irish study that prospectively documented reasons for ED visits in cancer patients who were under the care of a specialist palliative care program, about half of the visits were felt to be avoidable [73]. To a certain extent, the concept of avoidable ED visits is highly dependent on the alternative places available for care delivery. In the absence of readily available access to care elsewhere, ED visits may not be avoidable at all.

In summary, cancer patients at the EOL visit the ED because of symptoms related to their malignancy. Pain and respiratory issues are common, as are difficulties coping at home. Chemotherapy toxicity is a problem for all patients, regardless of their trajectory, and contributes significantly to ED visits.

Factors Associated with Increased Risk of ED Visits

ED visits at EOL have been shown to be associated with several different patient, tumor, treatment, and health system factors. However, the differences in definitions and populations studied lead to inconsistent results in the literature.

Among patient factors, sex has been most consistently demonstrated as an important factor with men more likely to visit the ED than women [8, 39, 41, 49, 52, 74–77]. Age is also important with older patients less likely to make visits [8, 39–41, 45, 49, 52, 75, 77–79]. Those with more significant comorbidity are also more likely to visit the ED [8, 41, 63, 74, 75, 80]. This is due to the complications related to chronic comorbidities, such as an exacerbation of COPD symptoms [81]. With one exception [40], those living in rural regions are also more likely to visit the ED [41, 52, 74–76, 78, 82]. ED visits are also more likely for patients who live in lower-income neighborhoods [52, 74, 76], with some exceptions [75]. Race has also been shown to play a factor [63].

Some tumor and treatment factors have been examined. Hematologic patients are more likely to make ED visits than patients with solid tumors [41, 79]. Among solid tumors, those with lung cancer are at the highest risk [74, 76, 77]. Patients with higher symptom burden are more likely to visit the ED [61]. Treatment factors are also important. Patients with metastatic cancer receiving chemotherapy are more likely to visit the ED [83]. As will be reviewed in the next section, palliative care interventions may decrease the likelihood of ED visits.

The influence of healthcare system factors has also been demonstrated. For example, as previously mentioned, harmonized comparisons between countries give some insight into the possible effects of different systems, such as the US hospice system versus more generalized access to palliative care [43]. Similarly, within the USA, those within the Department of Veterans Affairs (VA) system were less likely to have ED visits than those enrolled in Medicare [46].

Evidence for Strategies to Mitigate ED Use at EOL

It is unrealistic to expect that all ED visits at the EOL can be avoided. An ED visit can be a very taxing ordeal for patients at this point in their cancer journey, and most agree it happens more often than is desirable. This section will review the evidence for interventions that mitigate and reduce visits to the ED for patients at the EOL. A number of different studies have found that having access to palliative care programs can reduce ED visits for patients with advanced cancer at the end of life. For example, Temel et al.'s study in a comprehensive cancer care, where patients with advanced non-small cell lung cancer were randomized to early palliative care versus usual care, showed that early referral could reduce ED visits. Those randomized to the intervention arm experienced fewer ED visits in the last 30 days of life (22%) versus 30%) [84]. Evidence also exists for community-based palliative care interventions to reduce ED visits. A randomized controlled trial of home-based palliative care versus usual care also demonstrated a decrease in ED visits for those in the intervention arm, from 33% to 20% [85]. In addition, a pooled matched analysis of a retrospective cohort that was exposed to a specialized palliative home care team or usual home care demonstrated that the risk of having an ED visit at the EOL was 32% less for those who received care from the specialized palliative care teams [86]. The impact of early palliative care referral has also been demonstrated in observational datasets. For example, an Australian cohort of 28,331 decedents residing in an area with community-based palliative care program found that those who had the program initiated (58% of cohort) had less unplanned ED visits in the last 6 months of life [87].

Another study that showed access to palliative care services reduced ED visits was conducted as part of the Cancer Care Outcomes Research and Surveillance Consortium. This was a prospective cohort study of 1231 patients with stage IV lung or colorectal cancer, which demonstrated that patients who had an EOL discussion more than 30 days prior to their death were much less likely to use acute care services (ED and hospitalization) in the last 30 days [88]. A single-institution retrospective chart review of 220 women who died of ovarian cancer found that when an EOL discussion occurred more than 30 days prior to death, visits to the ED decreased [89].

Palliative care programs work because they can address many different factors of serious illnesses. Some of these include improved symptom control, decreased chemotherapy use, and improved knowledge of expectations. This was shown by Temel et al. in that patients in an intervention arm had better quality of life as well as less depression and anxiety [84]. Temel's study also revealed longer survival for those receiving palliative care. Palliative care is often provided by specialized care teams that are interdisciplinary, providing more opportunity for education and symptom management. They also tend to provide home-based services and are available 24/7. Nonetheless, some of the challenges to integrating palliative care into oncologic ED are the delays in palliative care consultations, patient and family expectations, and the fast-paced ED culture with limited resources [81].

While palliative care programs are important to reduce and avoid unnecessary ED visits, what can be done for those patients who do visit the ED? One intervention used a brief negotiated interview which allowed ED clinicians to motivate patients to have their needs met [90]. Another option for managing palliative care patients in the ED is to implement a screener questionnaire to identify patient needs, including cancer patients. In one study, a screener was integrated in the hospital's electronic health record system, which ultimately triggered referrals to consultants based on the results of the questions, e.g., pharmacy consult for medication reconciliation or social work consult for unmet needs at home [91]. This study showed that novel pathways and integrated communication led to many of the patients having their needs met quickly and efficiently. Another example of an intervention used in the ED is the Choosing Wisely campaign, which urged ED physicians to proactively engage with palliative care and hospice care services. One study used the Choosing Wisely campaign to create a pathway from ED to more appropriate services [92]. The ED physicians in the study would contact the palliative care team directly during regular office hours and allow them to decide next steps or, if during weekends or evenings, patients would be referred to a "hot clinic." The "hot clinic" saw patients in a timely manner, usually within a week of the ED visit. Another study in South Korea developed a dedicated ED cancer unit to specifically manage oncologic emergencies [60].

However, not all palliative care intervention studies in the ED have been positive [93–95]. Some studies have been positive for other outcomes, such as satisfaction, cost, or hospitalization, but not for ED visits [93, 96, 97]. For example, Bakitas et al. published a palliative care randomized trial using a nursing-led, multicomponent, psychoeducational intervention in a comprehensive cancer center [93]. This trial showed improvements to quality of life and mood, but did not reduce ED visits at the end of life. In a systematic review of the literature, DiMartino et al. could not substantiate whether palliative care interventions were more effective than usual care at reducing ED visits [98]. The nature of interventions and the different ways of reporting outcomes are perhaps a few reasons for the differing results.

It is important to recognize some of the factors that may explain varying outcomes in studies of ED visit mitigation. Intensity of care is important to note as a possible factor. In a study of patients receiving palliative home nursing, Seow et al. demonstrated a dose response such that patients receiving more nursing hours of care at home were less likely to visit the ED [75]. Studies conducted in two different provinces in Canada have demonstrated that increased continuity of physician care in the outpatient setting also decreased ED visits [52, 76]. Another factor may be a patient's access to hospice services. In one study, patients using hospice services were compared to matched cases who did not [36]. Patients enrolled in hospice services had less ED visits than their matched controls.

Structural aspects, for example, of a palliative care team, the health system, or the care setting itself, are also important. A case series of four regions with either high or low ED visit rates demonstrated that regions with lower ED rates have specific features of their palliative care systems that were absent in the other regions [99]. These included overall palliative care needs planning, a common chart, standardized patient assessments, 24/7 palliative care team access, advance practice nursing expertise available, and designated roles for the provision of palliative care services. Related to the physical structure of care settings, some jurisdictions have restructured the ED to create cancer-specific ED programs as an alternative place for assessment [60]. This particular structural change would facilitate easier access with short waits for cancer patients. It would also potentially hone the expertise of the ED staff working there and improve communication with the ambulatory team. Such an approach, however, seems to result from failure to optimize care upstream in the trajectory.

Future Directions

The literature provides some opportunities for better providing EOL care in the ED, when these unavoidable visits happen. These include providing education to ED clinicians on

the holistic nature of end-of-life care and educating staff on enhancing their communication techniques [10]. There is also opportunity for future research in this area. There has been little to no research evaluating the impact of psychoeducational interventions for informal caregivers on ED visits at EOL. A systematic review of informal caregivers' needs identified a lack of practical support for nursing skills [100]. The clinical approach taken for patients at EOL may be quite different [101] and may require additional training and skills. A systematic review of qualitative and quantitative studies of unmet needs for patients and carers identified that the most frequently unmet need was effective communication with healthcare professionals [102]. Another future opportunity is further exploration of hospital integration of palliative care divisions within EDs, which could lead to better coordination with existing services such as social workers, psychologists, or caseworkers [81].

Conclusion

Most practitioners are familiar with the aphorism, "To cure sometimes, to relieve often, to comfort always." The field of medicine has excelled in technical aspects of providing care, and we cure much more often than we did in the past. As a result, the importance of offering relief and comfort is sometimes forgotten. Population-based measures of ED visits in cancer patients at the EOL are a meaningful indicator of quality of care. With administrative healthcare data, this measure is easy to follow. Currently, ED visits are happening more often than is desirable. Increased efforts are needed to minimize the use of toxic therapies at the EOL and to create effective palliative care structures and processes which are readily accessible.

References

- Seow H, Barbera L, Sutradhar R, Howell D, Dudgeon D, Atzema C, et al. Trajectory of performance status and symptom scores for patients with cancer during the last six months of life. J Clin Oncol. 2011;29(9):1151–8.
- 2. World Health Organization. WHO definition of palliative care. http://www.who.int/cancer/palliative/definition/en/. Accessed 8 Oct 2020.
- Murray SA, Kendall M, Boyd K, Sheikh A. Illness trajectories and palliative care. BMJ. 2005;330(7498):1007–11.
- Ferrell BR, Temel JS, Temin S, Smith TJ. Integration of palliative care into standard oncology care: ASCO clinical practice guideline update summary. J Oncol Pract. 2017;13(2):119–21.
- Carr D, Goudas L, Lawrence D, Pirl W, Lau J, DeVine D, et al. Management of cancer symptoms: pain, depression, and fatigue. Evid Rep Technol Assess (Summ). 2002;(61):1–5.
- Bailey C, Murphy R, Porock D. Trajectories of end-of-life care in the emergency department. Ann Emerg Med. 2011;57(4):362–9.
- Earle CC, Park ER, Lai B, Weeks JC, Ayanian JZ, Block S. Identifying potential indicators of the quality of endof-life cancer care from administrative data. J Clin Oncol. 2003;21(6):1133–8.

- Earle CC, Neville BA, Landrum MB, Ayanian JZ, Block SD, Weeks JC. Trends in the aggressiveness of cancer care near the end of life. J Clin Oncol. 2004;22(2):315–21.
- 9. Forero R, McDonnell G, Gallego B, McCarthy S, Mohsin M, Shanley C, et al. A literature review on care at the end-of-life in the emergency department. Emerg Med Int. 2012;2012
- Gloss K. End of life care in emergency departments: a review of the literature. Emerg Nurse. 2017;25(2)
- Huang YL, Alsaba N, Brookes G, Crilly J. End-of-life care for older people in the emergency department: a scoping review. Emerg Med Australas. 2019;
- Kim K, Chakravarthy B, Anderson C, Liao S. To intubate or not to intubate: emergency medicine physicians' perspective on intubating critically ill, terminal cancer patients. J Pain Symp Manage. 2017;54(5):654–60 e1.
- Ouchi K, George N, Schuur JD, Aaronson EL, Lindvall C, Bernstein E, et al. Goals-of-care conversations for older adults with serious illness in the emergency department: challenges and opportunities. Ann Emerg Med. 2019;74(2):276–84.
- Brazil K, Howell D, Bedard M, Krueger P, Heidebrecht C. Preferences for place of care and place of death among informal caregivers of the terminally ill. Palliat Med. 2005;19(6)
- 15. Gomes B, Higginson IJ, Calanzani N, Cohen J, Deliens L, Daveson BA, et al. Preferences for place of death if faced with advanced cancer: a population survey in England, Flanders, Germany, Italy, the Netherlands, Portugal and Spain. Ann Oncol. 2012;23(8):2006–15.
- Higginson IJ, Sen-Gupta GJ. Place of care in advanced cancer: a qualitative systematic literature review of patient preferences. J Palliat Med. 2000;3(3)
- Stajduhar KI, Allan DE, Cohen SR, Heyland DK. Preferences for location of death of seriously ill hospitalized patients: perspectives from Canadian patients and their family caregivers. Palliat Med. 2008;22(1):85–8.
- Miyashita M, Morita T, Ichikawa T, Sato K, Shima Y, Uchitomi Y. Quality indicators of end-of-life cancer care from the bereaved family members' perspective in Japan. J Pain Symptom Manag. 2009;37(6):1019–26.
- Grunfeld E, Urquhart R, Mykhalovskiy E, Folkes A, Johnston G, Burge FI, et al. Toward population-based indicators of quality end-of-life care: testing stakeholder agreement. Cancer. 2008;112(10):2301–8.
- Wiese CH, Bartels UE, Ruppert D, Marung H, Luiz T, Graf BM, et al. Treatment of palliative care emergencies by prehospital emergency physicians in Germany: an interview based investigation. Palliat Med. 2009;23(4):369–73.
- 21. Marck CH, Weil J, Lane H, Weiland TJ, Philip J, Boughey M, et al. Care of the dying cancer patient in the emergency department: findings from a National survey of Australian emergency department clinicians. Intern Med J. 2014;44(4):362–8.
- 22. Reeve R, Srasuebkul P, Langton JM, Haas M, Viney R, Pearson SA, et al. Health care use and costs at the end of life: a comparison of elderly Australian decedents with and without a cancer history. BMC Palliat Care. 2017;17(1):1.
- Rosenwax LK, McNamara BA, Murray K, McCabe RJ, Aoun SM, Currow DC. Hospital and emergency department use in the last year of life: a baseline for future modifications to end-of-life care. Med J Aust. 2011;194(11):570–3.
- 24. Henson LA, Higginson IJ, Gao W. What factors influence emergency department visits by patients with cancer at the end of life? Analysis of a 124,030 patient cohort. Palliat Med. 2018;32(2):426–38.
- Public Health England. The National Archives. National End of Life Care Intelligence Network. Deaths from urological cancers in England, 2001–10. 10 Oct 2012. Accessed 8 Oct 2020.

- O'Hanlon C, Huang C, Sloss E, Anhang Price R, Hussey P, Farmer C, et al. Comparing VA and non-VA quality of care: a systematic review. J Gen Intern Med. 2017;32(1):105–21.
- American Society of Clinical Oncology. QOPI 2019 reporting tracks. https://practice.asco.org/quality-improvement/qualityprograms/quality-oncology-practice-initiative/qopi-relatedmeasures. Accessed 8 Oct 2020.
- Banerjee T, Mendler JH, Badri N, Hettler D, Berkhof JA, Abbas KJ, et al. Evaluating end-of-life (EOL) care at the University of Rochester Medical Center (URMC) using quality oncology practice initiative (QOPI) metrics (abstract). J Clin Oncol. 2019;37:27(Suppl):12–12.
- National Quality Forum. Palliative and end-of-life care 2015– 2016. Washington, DC; 2016.
- Ontario Health. Cancer Care Ontario. Cancer Quality Council of Ontario. End-of-life care measures 2016. https://www.csqi.on.ca/.
- Canadian Partnership Against Cancer. Person-centred perspective indicators in Canada: a reference report. Palliative and end-of-life care. Toronto, ON, Canada; 2017.
- 32. Falchook AD, Dusetzina SB, Tian F, Basak R, Selvam N, Chen RC. Aggressive end-of-life care for metastatic cancer patients younger than age 65 years. J Natl Cancer Inst. 2017;109(9):djx028.
- 33. Seow H, Barbera L, Pataky R, Lawson B, O'Leary E, Fassbender K, et al. Does increasing home care nursing reduce emergency department visits at the end of life? A population-based cohort study of cancer decedents. J Pain Symptom Manag. 2016;51(2):204–12.
- 34. Allende-Pérez S, González P, Peña-Nieves A, Herrera-Gómez Á, Verástegui E. End-of-life admission to the emergency department: experience of Mexico's national cancer institute of a developing country. Am J Hosp Palliat Care. 2020;37(11):881–4.
- 35. Alsirafy SA, Raheem AA, Al-Zahrani AS, Mohammed AA, Sherisher MA, El-Kashif AT, et al. Emergency department visits at the end of life of patients with terminal cancer: pattern, causes, and avoidability. Am J Hospice Palliat Care. 2016;33(7):658–62.
- 36. Obermeyer Z, Clarke AC, Makar M, Schuur JD, Cutler DM. Emergency care use and the medicare hospice benefit for individuals with cancer with a poor prognosis. J Am Geriatr Soc. 2016;64(2):323–9.
- 37. Lee YH, Chu D, Yang NP, Chan CL, Cheng SP, Pai JT, Chang NT. Emergency visits among end-of-life cancer patients in Taiwan: a nationwide population-based study. BMC Palliat Care. 2015;14:25.
- McNamara BA, Rosenwax LK, Murray K, Currow DC. Early admission to community-based palliative care reduces use of emergency departments in the ninety days before death. J Palliat Med. 2013;16(7):774–9.
- Miesfeldt S, Murray K, Lucas L, Chang CH, Goodman D, Morden NE. Association of age, gender, and race with intensity of endof-life care for Medicare beneficiaries with cancer. J Palliat Med. 2012;15(5):548–54.
- Maddison AR, Asada Y, Burge F, Johnston GW, Urquhart R. Inequalities in end-of-life care for colorectal cancer patients in Nova Scotia. Can J Palliat Care. 2012;28(2):90–6.
- Ho TH, Barbera L, Saskin R, Lu H, Neville BA, Earle CC. Trends in the aggressiveness of end-of-life cancer care in the universal health care system of Ontario. Can J Clin Oncol. 2011;29(12):1587–91.
- Saito AM, Landrum MB, Neville BA, Ayanian JZ, Weeks JC, Earle CC. Hospice care and survival among elderly patients with lung cancer. J Palliat Med. 2011;14(8):929–39.
- Warren JL, Barbera L, Bremner KE, Yabroff KR, Hoch JS, Barrett MJ, et al. End-of-life care for lung cancer patients in the United States and Ontario. J Natl Cancer Inst. 2011;103(11):853–62.
- 44. Bergman J, Saigal CS, Lorenz KA, Hanley J, Miller DC, Gore JL, et al. Hospice use and high-intensity care in men dying of prostate cancer. Arch Intern Med. 2011;171(3):204–10.

- Barbera L, Elit L, Krzyzanowska M, Saskin R, Bierman AS. End of life care for women with gynecologic cancers. Gynecol Oncol. 2010;118(2):196–201.
- 46. Keating NL, Landrum MB, Lamont EB, Earle CC, Bozeman SR, McNeil BJ. End-of-life care for older cancer patients in the Veterans Health Administration versus the private sector. Cancer. 2010;116(15):3732–9.
- 47. Setoguchi S, Glynn RJ, Stedman M, Flavell CM, Levin R, Stevenson LW. Hospice, opiates, and acute care service use among the elderly before death from heart failure or cancer. Am Heart J. 2010;160(1):139–44.
- Barbera L, Taylor C, Dudgeon D. Why do patients with cancer visit the emergency department near the end of life? CMAJ. 2010;182(6):563–8.
- 49. Tang ST, Wu SC, Hung YN, Chen JS, Huang EW, Liu TW. Determinants of aggressive end-of-life care for Taiwanese cancer decedents, 2001 to 2006. J Clin Oncol. 2009;27(27):4613–8.
- 50. Smith AK, Fisher J, Schonberg MA, Pallin DJ, Block SD, Forrow L, et al. Am I doing the right thing? Provider perspectives on improving palliative care in the emergency department. Ann Emerg Med. 2009;54(1):86–93, 93.e1.
- Barbera L, Paszat L, Qiu F. End-of-life care in lung cancer patients in Ontario: aggressiveness of care in the population and a description of hospital admissions. J Pain Symptom Manag. 2008;35(3):267–74.
- Burge F, Lawson B, Johnston G. Family physician continuity of care and emergency department use in end-of-life cancer care. Med Care. 2003;41(8):992–1001.
- Bach PB, Schrag D, Begg CB. Resurrecting treatment histories of dead patients: a study design that should be laid to rest. JAMA. 2004;292(22):2765–70.
- Earle CC, Ayanian JZ. Looking back from death: the value of retrospective studies of end-of-life care. J Clin Oncol. 2006;24(6):838–40.
- Earle CC, Neville BA, Landrum MB, Souza JM, Weeks JC, Block SD, et al. Evaluating claims-based indicators of the intensity of end-of-life cancer care. Int J Qual Health Care. 2005;17(6):505–9.
- Cowall DE, Yu BW, Heineken SL, Lewis EN, Chaudhry V, Daugherty JM. End-of-life care at a community cancer center. J Oncol Pract. 2012;8(4):e40–4.
- Barbera L, Seow H, Sutradhar R, Chu A, Burge F, Fassbender K, et al. Quality indicators of end-of-life care in patients with cancer: what rate is right? J Oncol Pract. 2015;11(3):e279–87.
- 58. Seow H, Arora A, Barbera L, McGrail K, Lawson B, Burge F, et al. Does access to end-of-life homecare nursing differ by province and community size?: a population-based cohort study of cancer decedents across Canada. Health Policy. 2018;122(2):134–9.
- Mayer DK, Travers D, Wyss A, Leak A, Waller A. Why do patients with cancer visit emergency departments? Results of a 2008 population study in North Carolina. J Clin Oncol. 2011;29(19):2683–8.
- Ahn S, Lee YS, Lim KS, Lee JL. Emergency department cancer unit and management of oncologic emergencies: experience in Asan Medical Center. Support Care Cancer. 2012;20(9):2205–10.
- 61. Barbera L, Atzema C, Sutradhar R, Seow H, Howell D, Husain A, et al. Do patient-reported symptoms predict emergency department visits in cancer patients? A population-based analysis. Ann Emerg Med. 2013;61(4):427–37. e5
- Vandyk AD, Harrison MB, Macartney G, Ross-White A, Stacey D. Emergency department visits for symptoms experienced by oncology patients: a systematic review. Support Care Cancer. 2012;20(8):1589–99.
- Lash RS, Bell JF, Reed SC, Poghosyan H, Rodgers J, Kim KK, et al. A systematic review of emergency department use among cancer patients. Cancer Nurs. 2017;40(2):135–44.
- Caterino JM, Adler D, Durham DD, Yeung SJ, Hudson MF, Bastani A, et al. Analysis of diagnoses, symptoms, medications,

and admissions among patients with cancer presenting to emergency departments. JAMA Netw Open. 2019;2(3):e190979.

- 65. Leak A, Mayer DK, Wyss A, Travers D, Waller A. Why do cancer patients die in the emergency department? An analysis of 283 deaths in NC EDs. Am J Hospice Palliat Med. 2013;30(2):178–82.
- 66. Yildirim B, Tanriverdi O. Evaluation of cancer patients admitted to the emergency department within one month before death in Turkey: what are the problems needing attention? Asian Pac J Cancer Prev. 2014;15(1):349–53.
- 67. Verhoef MJ, de Nijs E, Horeweg N, Fogteloo J, Heringhaus C, Jochems A, et al. Palliative care needs of advanced cancer patients in the emergency department at the end of life: an observational cohort study. Support Care Cancer. 2020;28(3):1097–107.
- 68. Yang GM, Ng E, Lian S, Ong MEH. Reasons for emergency department visits among advanced cancer patients in their last week of life. Proc Singapore Healthc. 2018;27(1):59–62.
- 69. Fisch MJ, Lee JW, Weiss M, Wagner LI, Chang VT, Cella D, et al. Prospective, observational study of pain and analgesic prescribing in medical oncology outpatients with breast, colorectal, lung, or prostate cancer. J Clin Oncol. 2012;30(16):1980–8.
- Barbera L, Seow H, Husain A, Howell D, Atzema C, Sutradhar R, et al. Opioid prescription after pain assessment: a populationbased cohort of elderly patients with cancer. J Clin Oncol. 2012;30(10):1095–9.
- Deandrea S, Montanari M, Moja L, Apolone G. Prevalence of undertreatment in cancer pain. A review of published literature. Ann Oncol. 2008;19(12):1985–91.
- Hwang U, Richardson L, Livote E, Harris B, Spencer N, Sean MR. Emergency department crowding and decreased quality of pain care. Acad Emerg Med. 2008;15(12):1248–55.
- Wallace EM, Cooney MC, Walsh J, Conroy M, Twomey F. Why do palliative care patients present to the emergency department? Avoidable or unavoidable? Am J Hospice Palliat Care. 2013;30(3):253–6.
- Barbera L, Paszat L, Chartier C. Indicators of poor quality end-oflife cancer care in Ontario. J Palliat Care. 2006;22(1):12–7.
- 75. Seow H, Barbera L, Howell D, Dy SM. Using more end-oflife homecare services is associated with using fewer acute care services: a population-based cohort study. Med Care. 2010;48(2):118–24.
- 76. Almaawiy U, Pond GR, Sussman J, Brazil K, Seow H. Are family physician visits and continuity of care associated with acute care use at end-of-life? A population-based cohort study of homecare cancer patients. Palliat Med. 2014;28(2):176–83.
- Krzyzanowska MK, Barbera L, Elit L, Kwon J, Lofters A, Saskin R, Yeritsyan N, Bierman AS. Cancer. In: Bierman AS, editor. Project for an Ontario women's health evidence-based report, vol. 1. Toronto, Canada; 2009.
- Lawson BJ, Burge FI, McIntyre P, Field S, Maxwell D. Palliative care patients in the emergency department. J Palliat Care. 2008;24(4):247–55.
- Hui D, Didwaniya N, Vidal M, Shin SH, Chisholm G, Roquemore J, et al. Quality of end-of-life care in patients with hematologic malignancies: a retrospective cohort study. Cancer. 2014;120(10):1572–8.
- Quon H, Shepherd FA, Payne DG, Coy P, Murray N, Feld R, et al. The influence of age on the delivery, tolerance, and efficacy of thoracic irradiation in the combined modality treatment of limited stage small cell lung cancer. Int J Radiat Oncol Biol Phys. 1999;43(1):39–45.
- Elsayem AF, Elzubeir HE, Brock PA, Todd KH. Integrating palliative care in oncologic emergency departments: challenges and opportunities. World J Clin Oncol. 2016;7(2):227–33.
- Brazil K, Kaasalainen S, Williams A, Rodriguez C. Comparing the experiences of rural and urban family caregivers of the terminally ill. Rural Remote Health. 2013;13(1):2250.

- Keam B, Oh DY, Lee SH, Kim DW, Kim MR, Im SA, et al. Aggressiveness of cancer-care near the end-of-life in Korea. Jpn J Clin Oncol. 2008;38(5):381–6.
- Temel JS, Greer JA, Muzikansky A, Gallagher ER, Admane S, Jackson VA, et al. Early palliative care for patients with metastatic non-small-cell lung cancer. N Engl J Med. 2010;363(8):733–42.
- 85. Brumley R, Enguidanos S, Jamison P, Seitz R, Morgenstern N, Saito S, et al. Increased satisfaction with care and lower costs: results of a randomized trial of in-home palliative care. J Am Geriatr Soc. 2007;55(7):993–1000.
- 86. Seow H, Brazil K, Sussman J, Pereira J, Marshall D, Austin PC, et al. Impact of community based, specialist palliative care teams on hospitalisations and emergency department visits late in life and hospital deaths: a pooled analysis. BMJ. 2014;348:g3496.
- 87. Wright CM, Youens D, Moorin RE. Earlier Initiation of community-based palliative care is associated with fewer unplanned hospitalizations and emergency department presentations in the final months of life: a population-based study among cancer decedents. J Pain Symptom Manage. 2018;55(3):745–54. e8.
- Mack JW, Cronin A, Keating NL, Taback N, Huskamp HA, Malin JL, et al. Associations between end-of-life discussion characteristics and care received near death: a prospective cohort study. J Clin Oncol. 2012;30(35):4387–95.
- Lopez-Acevedo M, Havrilesky LJ, Broadwater G, Kamal AH, Abernethy AP, Berchuck A, et al. Timing of end-of-life care discussion with performance on end-of-life quality indicators in ovarian cancer. Gynecol Oncol. 2013;130(1):156–61.
- 90. Leiter RE, Yusufov M, Hasdianda MA, Fellion LA, Reust AC, Block SD, et al. Fidelity and feasibility of a brief emergency department intervention to empower adults with serious illness to initiate advance care planning conversations. J Pain Symptom Manag. 2018;56(6):878–85.
- Reuter Q, Marshall A, Zaidi H, Sista P, Powell ES, McCarthy DM, et al. Emergency department-based palliative interventions: a novel approach to palliative care in the emergency department. J Palliat Med. 2019;22(6):649–55.
- 92. Koh MYH, Lee JF, Montalban S, Foo CL, Hum AYM. ED-PALS: a comprehensive palliative care service for oncology patients

in the emergency department. Am J Hospice Palliat Care. 2019;36(7):571-6.

- 93. Bakitas M, Lyons KD, Hegel MT, Balan S, Brokaw FC, Seville J, et al. Effects of a palliative care intervention on clinical outcomes in patients with advanced cancer: the Project ENABLE II randomized controlled trial. JAMA. 2009;302(7):741–9.
- 94. Aiken LS, Butner J, Lockhart CA, Volk-Craft BE, Hamilton G, Williams FG. Outcome evaluation of a randomized trial of the PhoenixCare intervention: program of case management and coordinated care for the seriously chronically ill. J Palliat Med. 2006;9(1):111–26.
- Rabow MW, Dibble SL, Pantilat SZ, McPhee SJ. The comprehensive care team: a controlled trial of outpatient palliative medicine consultation. Arch Intern Med. 2004;164(1):83–91.
- Zimmer JG, Groth-Juncker A, McCusker J. A randomized controlled study of a home health care team. Am J Public Health. 1985;75(2):134–41.
- Hughes SL, Cummings J, Weaver F, Manheim L, Braun B, Conrad K. A randomized trial of the cost effectiveness of VA hospital-based home care for the terminally ill. Health Serv Res. 1992;26(6):801–17.
- 98. DiMartino LD, Weiner BJ, Mayer DK, Jackson GL, Biddle AK. Do palliative care interventions reduce emergency department visits among patients with cancer at the end of life? A systematic review. J Palliat Med. 2014;17(12):1384–99.
- 99. Sussman J, Barbera L, Bainbridge D, Howell D, Yang J, Husain A, Librach SL, Viola R, Walker H. Health system characteristics of quality care delivery: a comparative case study examination of palliative care for cancer patients in four regions in Ontario. Can Palliat Med. 2012;26(4):322–35.
- 100. Bee PE, Barnes P, Luker KA. A systematic review of informal caregivers' needs in providing home-based end-of-life care to people with cancer. J Clin Nurs. 2009;18(10):1379–93.
- Nauck F, Alt-Epping B. Crises in palliative care--a comprehensive approach. Lancet Oncol. 2008;9(11):1086–91.
- 102. Ventura AD, Burney S, Brooker J, Fletcher J, Ricciardelli L. Home-based palliative care: a systematic literature review of the self-reported unmet needs of patients and carers. Palliat Med. 2014;28(5):391–402.



Pain Management

Danielle M. McCarthy and Laurie M. Aluce

Case Study

A 56-year-old man with a history of metastatic pancreatic cancer receiving gemcitabine chemotherapy presents to the ED with 2 days of worsening abdominal pain. He describes the pain as sharp, constant, located throughout his entire abdomen, and ranks it an 8/10 on the numerical pain rating scale. He also feels nauseated and has not had a bowel movement in 2 days. He denies fever or urinary symptoms.

He states that his cancer-related chronic pain is usually well controlled with his fentanyl patch and extended release morphine, so he usually only needs a couple doses of his PRN immediate-release morphine per day. In the last 24 hours, however, he has taken the immediate-release morphine exactly every 4 hours due to pain.

The nurse asks the physician, "What would you like to give for pain?" The physician is torn. The etiology of the pain remains unclear, and he does not want to make any potential opioid-induced constipation worse. The patient is in a severe pain crisis, however, so the physician decides to treat with IV opioids while awaiting imaging results. He orders 6 mg IV morphine.

After 15 minutes, the patient remains in severe, 8/10 pain despite the morphine. He is not exhibiting any adverse side effects from the opioids. The physician wonders if he should order a higher dose of morphine. After consulting his equianalgesic table and converting the patient's home pain medication requirements into IV morphine equivalents, the physician realizes he severely under-dosed the first round of morphine. The physician orders a second, more appropriate dose of IV morphine, which provides adequate relief to the patient.

D. M. McCarthy () · L. M. Aluce ()

Department of Emergency Medicine, Northwestern Memorial Hospital, Feinberg School of Medicine, Northwestern University, Chicago, IL, USA e-mail: d-mccarthy2@northwestern.edu;

laurie.nosbusch@northwestern.edu

Introduction

Pain is one of the most common symptoms experienced by patients with cancer. Prevalence estimates (pooling patients with multiple types of cancer and in various stages of treatment) suggest that >50% of cancer patients experience pain [1]. The prevalence of pain is higher for patients who have metastatic or advanced stage disease, and greater than a third of cancer patients rate their pain as moderate or severe [1]. The prevalence of pain in the subset of cancer patients visiting the emergency department is less well defined. It is estimated that approximately 10-41% of all ED visits made by cancer patients involve pain as the primary complaint [2, 3]. Regardless of the chief complaint that prompted the ED visit (e.g., presenting primarily for pain crisis or presenting for infectious symptoms), giving analgesics is among the most frequent treatments provided to cancer patients in the ED. The severity of patient self-reported symptoms, including pain, is a reliable predictor of emergency department visits [4, 5].

Classification of Pain

The National Comprehensive Cancer Network states that a "pain crisis" is an occasion on which a patient is experiencing severe pain (at least a numerical rating of 8 on a 10-point scale) [6]. Such a pain crisis may arise from inadequate control of a patient's baseline persistent pain or may be a more acute pain crisis superimposed on a well-established history of chronic pain. Such pain is commonly referred to as "breakthrough pain." While breakthrough pain may result from cancer progression, it may also result from increased analgesic tolerance or end-of-dose failure. Similar to the more general pain prevalence estimates detailed above, nearly two-thirds of patients with chronic cancer pain syndromes experience breakthrough pain episodes [7]. In general, it is accepted that a patient's baseline persistent pain must be well controlled before attributing the pain episode to "break-through pain" [8].

Such pain crises require a rapid response from emergency physicians to achieve adequate analgesia. The approach to a pain crisis should be similar to the emergency physician's approach to any other emergency and include an assessment of the history, severity, timing, and location of the pain, in addition to a thorough physical exam. In addition to classifying pain in cancer patients as baseline pain or breakthrough pain, other important considerations when assessing pain in cancer patients include the following: awareness of the pain mechanism or pathophysiology, consideration of the triggers of the pain crisis, and discussion of the goals of care.

Pain Severity

Many tools are available to assess pain severity and thereby quantify pain. Examples of such pain severity assessment tools include numerical rating scales (NRS), visual analogue scales (VAS), and picture scales (e.g., Faces Pain Scale). More in-depth assessments, which often incorporate numerical or visual analogue rating scales, are also available and include the Brief Pain Inventory, the McGill Pain Ouestionnaire, and the Memorial Pain Assessment Card. Pain is subjective, and at times in the ED, high pain scores are greeted with skepticism; therefore, it is important to use one of these validated measures to assess patients' pain. This is particularly important for patients with chronic pain who may not exhibit more objective signs (e.g., grimacing) or vital sign changes such as tachycardia [9]. Using a pain score measure, and applying it consistently, will help to add a more objective means of reproducibly tracking patients' pain and response to treatment over time. Although any of the abovementioned measures are validated and can be useful if applied consistently, for purposes of ED evaluation of cancer pain, a numerical rating scale (NRS) is preferred [10, 11]. Specifically, in cancer patients, a NRS has better capability to distinguish between a patient's background or chronic pain and breakthrough pain [11]. The NRS is an 11-point scale ranging from 0 to 10. When asking patients to rate their pain on this scale, it is important to provide consistent verbal anchors to the scale [12]. Most commonly, the anchors are "0 = no pain" and "10 = worst pain possible" or "10 = pain as intense as you can imagine" [10].

In addition to pain ratings, it is important to ask patients their expectations regarding pain relief. Asking "At what level of pain do you feel comfortable?" recognizes that patients with chronic pain, including those with chronic cancer pain, do not necessarily expect pain intensity scores to reach zero [13]. Use of such personalized pain goals may allow adequate analgesia while avoiding the overtreatment of pain and resultant adverse effects.

Pain Mechanisms

In addition to assessing pain intensity, the pathophysiology and trigger of the pain should be considered. No universally accepted system for classifying cancer pain exists [14]. Nonetheless, cancer pain is often described in terms of the pathophysiology of the pain and is broadly divided into nociceptive or neuropathic pain. Nociceptive pain is caused by tissue injury and can be further subdivided into visceral pain (from organs) or somatic pain (related to bones, joints, soft tissue). Visceral pain (e.g., peritoneal carcinomatosis) is more poorly localized than somatic pain (e.g., bone metastases). Neuropathic pain is related to dysfunction of nerves; this dysfunction may be central or peripheral and has many possible etiologies, including direct compression of a nerve or related to treatment (e.g., inflammation post-radiation). Often, those with cancer suffer from complex pain states combining both nociceptive and neuropathic mechanisms. Chronic pain in cancer survivors may have a unique etiology and symptomatology [6]. Studies have shown that chronic pain affects approximately one-third of cancer survivors and that, even among survivors who are disease free, approximately one-third continue to use opioids long term [15, 16]. Considering the pathophysiology of the pain is important because different types of pain may respond to treatments differently. Additionally, patients with cancer may suffer from non-cancer-related pain syndromes such as pre-existing diabetic neuropathy or arthritis pain. Although these pain syndromes are not directly related to the cancer, they can nonetheless contribute to patients' psychological distress and suffering.

Other historical factors, such as recent treatments and the stage of the cancer, can also aid the emergency physician in defining a trigger for the pain. This information can help the physician determine if the pain represents a reversible pain crisis, an anticipated worsening related to recent treatment, or a worsening related to progression of disease. A new pain or pain in a new location may represent disease progression and as such may require more extensive diagnostic evaluation than an increase in intensity of a known chronic pain.

Before beginning an extensive diagnostic search for the cause of a new pain and before starting a patient on a new analgesic, it is important to discuss the goals of care with the patient as part of a shared decision approach to care. These conversations can be difficult because of the lack of a pre-existing relationship, but are nonetheless important. The extent of the patient's diagnostic evaluation will depend on their goals of care and should take into consideration the risk or discomfort of diagnostic tests and what action would be taken with different results of those tests.

Oligoanalgesia

Any discussion of pain control in emergency medicine should include a discussion of oligoanalgesia - the underuse of analgesics. Oligoanalgesia has been a problem for cancer patients but is perhaps improving in recent years. A recent review found that the percentage of cancer patients with undertreated pain decreased from 43% to 32% over the course of 6 years [17]. This estimate implies that, while there has been an improvement in the treatment of pain in cancer patients, many patients seeking care in the ED for breakthrough pain may still have had inadequate baseline pain control. In fact, one recent study found that opioid-tolerant cancer patients were less likely to receive an adequate initial pain medication dose in the ED if they had higher home opioid doses [18]. Psychosocial factors such as depression, spiritual concerns, or misconceptions regarding prescribed medications may also contribute to oligoanalgesia in cancer patients [19]. The problem of oligoanalgesia for these patients is then compounded by a well-established history of oligoanalgesia in the ED itself. Specifically, recent studies have found that pain is undertreated in the ED and that disparities in pain treatment exist related to age and race [20-23]. Many factors likely contribute to oligoanalgesia in the ED. One concern in particular relates to significant tension between providing adequate analgesia and ongoing concerns relating to drug misuse, addiction, and deaths from prescription opioids in the USA [24]. According to the Centers for Disease Control and Prevention, the overall opioid prescribing rate decreased 19.2% between 2006 and 2017, with the largest annual reductions occurring in the most recent 4 years [25]. A study regarding opioid prescribing practices between 2007 and 2013 showed that opioid prescribing stayed relatively constant for cancer patients while it increased by approximately 2% for non-cancer patients [26]. Further research will need to be conducted to monitor these trends and watch for possible decreases in opioid prescriptions to cancer patients as the opioid epidemic continues and overall opioid prescribing declines. In the context of the cancer patient in the ED, the immediate focus should be on controlling the breakthrough pain during the ED visit. Decisions about new prescription analgesics from the ED can be made in conjunction with the treating oncologist or palliative care physician.

Treatment Options

Treatment of cancer pain should be individualized. There are many treatment modalities available to the cancer patient, including pharmacologic, interventional (e.g., injection therapies, neural blockade), rehabilitative (e.g., therapy for lymphedema), psychological (e.g., cognitive behavioral therapy), neurostimulation, or integrative (e.g., massage) [27, 28]. Those modalities may be applied singularly or in a combined modality fashion. Most of these therapies are not immediately accessible to the emergency physician; thus, while an awareness of these therapies is important, the primary tools at the emergency physician's disposal to improve pain are pharmacologic.

WHO Analgesic Ladder

Over 30 years ago, in 1986, the World Health Organization published recommendations for the management of cancer pain. These recommendations included a three-step analgesic "ladder" intended to guide the selection and escalation of analgesics [29] (Fig. 56.1). This model has been frequently used for not only cancer pain but other painful syndromes. Over the intervening 30 years, modifications to the existing ladder have been suggested, including eliminating the second step of the ladder, adding a fourth step, or "fast tracking" patients to the top of the ladder [30, 31]. When considering the individual patient, it is important to reconcile evidencebased clinical guidelines and new information (e.g., new medications, importance of risk assessment) with the original WHO consensus-based guidelines [27]. However, the WHO analgesic ladder provides a good framework for the

WHO's Pain Relief Ladder



Fig. 56.1 WHO analgesic ladder. (Reprinted with permission of The World Health Organization: http://www.who.int/cancer/palliative/pain-ladder/en/. Accessed 18 Mar 2020)

discussion of the different types of analgesics and the rationale that analgesic choice should be given commensurate to the patients' pain intensity, as measured by a pain scale.

Non-opioid Analgesics

Non-opioid analgesics include nonsteroidal antiinflammatory medications (NSAIDs) and acetaminophen. These medications are useful in the management of acute and chronic pain. One limitation of both NSAIDs and acetaminophen is a "ceiling effect" wherein increasing the dose above a certain level does not provide any additional pain relief. The ceiling effect therefore limits the ability of these medications to be titrated for severe pain; however, even when NSAIDs and acetaminophen alone are insufficient for pain control, they should be considered as a co-analgesic to opioid treatment because they may reduce the dose of opioid needed to achieve pain control.

Acetaminophen is a non-salicylate analgesic that does not have any antiplatelet activity. It does not have clinically detectable anti-inflammatory effects; however, it may be a useful analgesic for some conditions. One concern with acetaminophen is potential hepatotoxicity. The FDA advises that doses in patients with normal liver function should not exceed 4000 mg/day, and the National Comprehensive Cancer Network (NCCN) panel suggests that providers consider limiting chronic administration of acetaminophen to 3000 mg or less per day [6].

NSAID medications possess both anti-inflammatory and analgesic properties. These medications act by inhibiting isoforms of an enzyme called cyclooxygenase (COX); inhibition of this enzyme results in decreased synthesis of prostaglandins. Different classes of NSAIDs have varying selectivity for the isoforms of the COX enzyme (COX-1 and COX-2). Depending on the source of the patient's pain and comorbidities, a choice of a nonselective or selective NSAID can be made. Patients who respond to one of the NSAIDs may not respond as well to others. NSAID medications also have significant side effects that should be considered prior to administration. All NSAIDs cause a reversible decreased platelet aggregation (while the drug is at therapeutic serum concentrations). If patients already have a high risk of bleeding, the use of NSAIDs should be carefully considered. All NSAIDs can cause gastrointestinal (GI) adverse effects, ranging from dyspepsia to bleeding gastric ulcers. Medications to protect the gastric mucosa, such as a proton pump inhibitor (PPI), should be considered to minimize the risk. NSAIDs can also lead to renal insufficiency through multiple mechanisms, and caution should be used when patients have risk factors for renal impairment, including advanced age, dehydration, diuretic use or multiple myeloma.

Opioid Analgesics

When a patient's pain is not adequately controlled by nonopioid analgesics, opioid analgesics represent the next step on the WHO analgesic ladder. Opioid medications can be classified as naturally occurring opioids (e.g., morphine, codeine) and semisynthetic (e.g., dihydromorphone, oxycodone) or synthetic compounds (e.g., fentanyl, methadone). Opioids can further be classified by their action at the opioid receptor (agonist, partial agonist, or antagonist) and by the receptor where they primarily function (mu, delta, or kappa). Opioid receptors exist in both the central nervous system and the peripheral tissues; however, the clinical effects of opioids are thought to be related primarily to the opioid action on central rather than peripheral receptors [32].

Tramadol is a mixed-mechanism drug that acts as a weak mu opioid receptor agonist but also demonstrates some norepinephrine and serotonin reuptake inhibition. A recent Cochrane review concluded that there is little evidence to support the use of tramadol for cancer pain and that it is less effective than morphine [33].

Opioids can be administered by multiple means, including oral, transmucosal, rectal, transdermal, intranasal, subcutaneous, or intravenous routes. The intramuscular route is generally not recommended as it provides no pharmacologic advantage over subcutaneous administration and has the disadvantage of causing additional pain. If time permits, the oral route of administration is preferred [29]; however, for patients in a severe pain crisis, more rapid pain control may necessitate the use of intravenous opioids. Opioids, with the exception of methadone, follow first-order kinetics and achieve their peak plasma concentration (and maximal analgesic effect) along a similar timeline: 60–90 min for oral/ rectal administration, 30 min for subcutaneous/intramuscular administration, and 6–10 min for intravenous administration [19, 34].

If time permits the administration of an oral opioid, it is also important to remember that many oral opioids are combination pills with a non-opioid analgesic (e.g., acetaminophen). The presence of the non-opioid limits titration of the medication orally to avoid toxicity from the co-analgesic (e.g., maximum of 4000 mg/day of acetaminophen).

For those patients in more severe pain, or requiring intravenous dosing for other reasons (difficulty swallowing), there are many available intravenous opioids. For the opioidnaïve patient, morphine is a safe, standard drug to start therapy [6, 35]. However, morphine should be used with caution in patients with renal impairment because one of the active metabolites (morphine-6-glucoronide) can accumulate with renal dysfunction. For intravenous dosing of the opioid-naïve patient, a starting dose of 2–5 mg of morphine (or equivalent) is recommended. This dose should be followed by a reassessment at 15 min, and if the pain score remains

-	-	-	
Opioid	Oral dose	Parenteral dose	Duration of action (h)
Morphine	30 mg	10 mg	3–4
Hydrocodone	30 mg	-	4-8
Hydromorphone	7.5 mg	1.5 mg	3–4
Oxycodone	15-20 mg	-	3–6
Fentanyl ^b	-	50–100 µg	1-2

 Table 56.1
 Equianalgesic dosing^a

^aThis table is a guide only. Equianalgesic dosing tables vary subtly and this table should not replace more in-depth review of dosing

^bApplies to IV conversion only. For transdermal fentanyl conversion, see package insert

unchanged or increased, the initial dose given should be increased by 50-100%. If the pain score is decreased but still moderate (e.g., 4-7), the same initial dose should be repeated, and if the pain level is low (e.g., 0-3), then the initial dose can be used as needed [6].

For the opioid-tolerant patient (a patient that takes at least 60 mg/day oral morphine, 25 mcg/hr transdermal fentanyl, 30 mg/day oral oxycodone, 8 mg/day oral hydromorphone, 25 mg/day oral oxymorphone, or an equianalgesic dose of another opioid for 1 week or longer), the drug choice will likely be informed by their home medications and prior opioid use [6]. Patients may be on a combination of opioid medications at home (e.g., transdermal and oral preparations or long-acting and immediate-release preparations). In order to identify the approximate opioid use of a patient at home, and thereby to more accurately estimate their pain control needs in the ED, equianalgesic dosing tables can be used (Table 56.1).

Equianalgesic dosing tables were first constructed in the 1960s and 1970s to codify the relative potency of different opioid formulations. These tables can be used to calculate patients' baseline outpatient opioid use. Recently, experts have expressed concerns over the limitations of the data used to construct the dosing tables and their applicability in the clinical realm [36, 37]. Specifically, the studies used to construct these tables were primarily acute rather than chronic pain. Research also suggests that there is significant variability in the dose needed to achieve pain relief between individuals which may be influenced not only by prior exposure to opioids but also by age, gender, and genetic polymorphisms that affect opioid binding [9]. Given this potential for wide variability in dose response, the equianalgesic dosing recommendations as well as the starting doses noted above should be considered a guideline, to be individualized by the practicing physician.

When using the equianalgesic dosing tables, the first step is to calculate the patient's "equianalgesic dose equivalent" based on the analgesic use over the previous 24 h. After choosing the opioid to administer, the initial IV dose will be determined by converting the previous 24-h requirement to a total IV equivalent. Once a 24-h IV equivalent has been calculated, the first dose should be 10-20% of that total dose [6]. After the first dose is given, a reassessment should occur within 15 min, and, similar to opioid-naïve patients, if the pain level remains ≥ 8 , the dose should be escalated by 50-100%. If the pain is moderate (e.g., 4-7), the same initial dose should be repeated, and if the pain level is low (e.g., 0-3), then the initial dose can be used as needed [6].

Another reason to exercise caution when switching opioids is because of opioid cross-tolerance. Patients on opioids will develop tolerance; this is expected. One of the signs of tolerance is achieving less pain relief from the same dose of the medication. One approach to optimize pain control when tolerance is present is to switch (or rotate) the opioid being used. In other clinical situations, opioid rotation may be useful to minimize adverse effects. Although there is some cross-tolerance between opioids, it is not complete. This incomplete cross-tolerance is due to many factors, including individual variations in metabolism, concurrent medications that impact metabolisms, and individual variations in opioid receptors. Due to this variability in individual response when switching opioids, it is recommended to first calculate the equianalgesic dose and then to decrease the calculated dose by 25-50% [9, 36]. Ultimately, the clinical situation should be considered when choosing a 25% reduction vs. a 50% reduction, including factors such as the patient's pain control and individual adverse effect profile. Two medications with exceptions to the 25-50% automatic reduction are methadone and fentanyl. Methadone has a nonlinear relationship to other opioids [19]. When switching to methadone, larger automatic dose reductions are recommended (75-90%) [27]. Converting to transdermal fentanyl should follow the calculated equianalgesic dose in the package insert and does not require an automatic dose reduction [28]. Initiation and titration of methadone and fentanyl are complex and should only be done by emergency physicians in consultation with the treating oncologist, pain specialist, or palliative care team.

Opioid Side Effects

Opioid analgesics have many potential side effects that may make patients or prescribers reluctant to use the medications or use them in adequate doses to achieve pain control. Patients can develop a tolerance to certain side effects (e.g., nausea or respiratory depression); however, other side effects (e.g., constipation) are not decreased with chronic use (Table 56.2). Several symptoms including pruritus and rash may result from either allergy or direct opioid effects from mast cell degeneration and histamine release. True anaphylaxis to opioids is rare, but can occur. Care must also be taken to distinguish between opioid side effects and the underlying clinical manifestations of comorbidities such as dehydration or drug interactions [38].

Adverse effect	Develop tolerance	Treatment
Constipation	No	Use laxative prophylactically
Pruritus	No	Opioid rotation
Nausea	Often	Opioid rotation Anti-nausea medications
Sedation	Yes	Decrease or rotate opioid Discontinue other medications that can cause sedation
Respiratory depression	Yes	Decrease or rotate opioid Discontinue other medications that can cause sedation Sedation will precede respiratory depression
Delirium	No	Decrease or rotate opioid Discontinue other medications that can cause delirium Avoid sedating medications unless necessary and consider the use of antipsychotic medications
Hyperalgesia	No	Rotate opioid Use non-opioid medication/strategy

Table 56.2 Adverse effects of opioids

Respiratory depression is one of the most feared side effects of opioids; however, tolerance can develop in a period of days to weeks of being on opioids, and it is rare in patients taking opioids chronically. Respiratory depression occurs in a dose-dependent fashion due to opioid action at the brain stem respiratory centers. Concomitant use with other sedating medications, such as benzodiazepines, may increase the risk for respiratory depression. Sedation precedes respiratory depression. If there is a concern for respiratory depression (from either home medications or medications administered in the ED), naloxone can be administered. Naloxone is an opioid receptor antagonist which will reverse the effects of the opioid; however, the half-life of naloxone is shorter than the half-lives of many opioids, so patients need continued observation if there was a concern for respiratory depression. Naloxone can precipitate acute and severe withdrawal symptoms in the patient taking chronic opioids and should be administered cautiously. Diluting 1 mL of the standard 0.4 mg/mL concentration of naloxone in 9 mL of normal saline (for a total of 10 mL) and administering 1-2 mL (0.04–0.08 mg) approximately every minute until respiratory rate improves will allow reversal of respiratory depression while minimizing withdrawal symptoms.

Nausea is one of the most common side effects of opioids with estimates ranging from 10% to 40% of patients experiencing nausea [39]. Opioid causes nausea through several mechanisms including stimulation of the chemoreceptor trigger zone, slowed gastric emptying, and effects on the vestibular system. Depending on the source of the nausea, different antiemetics will have variable efficacy. Dopamine receptor antagonists (e.g., prochlorperazine or haloperidol) or serotonin antagonists (e.g., ondansetron) will be the most useful for nausea related to the chemoreceptor trigger zone, whereas promotility agents (e.g., metoclopramide) may have more impact when gastric stasis is causing the nausea.

Constipation is another common side effect that emergency physicians should anticipate in patients for whom they prescribe opioids. Among cancer patients on chronic opioids, the prevalence of constipation is as high as 90% [40]. Opioids slow bowel transit time and peristalsis, and tolerance to constipation does not develop over time. When a cancer patient presents with symptoms of constipation, it is important to rule out bowel obstruction/impaction before starting medications to treat constipation.

Pharmacologic agents to ease constipation are typically divided into five categories: bulk-forming agents, softeners, stimulants, osmotic agents, and peripheral mu opioid receptor antagonists. Bulk-forming agents increase fecal mass to stimulate peristalsis. Stimulants act by increasing intestinal motility, whereas osmotic agents (e.g., polyethylene glycol, lactulose) act by increasing water content in the large bowel. The NCCN recommends prophylaxis with both a stimulant (or pro-kinetic) agent and the osmotic agent polyethylene glycol [6]. Bulk-forming agents and stool softeners are unlikely to be effective in isolation. A 2010 Cochrane review recommended the use of polvethylene glycol over lactulose for chronic constipation because of better outcomes related to stool frequency, form, associated abdominal pain, and use of additional products [41]. If constipation persists despite the above medications, the provider can titrate the existing regimen or add an additional agent, such as magnesium hydroxide [6]. Two peripherally acting mu opioid receptor antagonists may be considered if laxative therapy has failed. Both methylnaltrexone, administered subcutaneously or orally, and naloxegol, an orally active agent, have demonstrated efficacy in reversing opioid-induced constipation [42, 43].

Adjuvants (Co-analgesics)

Adjuvant medications, also known as co-analgesics, are a diverse group of drugs that may have a primary indication other than pain; however, they work to enhance the effects of traditional analgesics, "have independent analgesic activity in certain painful conditions, or counteract the adverse effects of analgesics" [9, 44]. The emergency physician may not be starting these medications in the ED to achieve pain relief in the acute setting; however, it is important to have a familiarity with these medications both when taking the patient's history and when discussing future treatment options with their outpatient oncologist or palliative care team.

Adjuvant medications are on every step of the WHO pain ladder, and they encompass many drug classes, including

Category	Example	Indication	Comments
Corticosteroids	Dexamethasone Methylprednisolone	Spinal cord compression Bone metastases Neuropathic pain	Often used to treat emergencies associated with cancer progression (e.g., spinal cord compression) but also have utility for other painful conditions including bone metastases and neuropathic pain The use of corticosteroids should be discussed with the oncologist given possibility of affecting treatment course
Benzodiazepines	Lorazepam Diazepam	Anxiety Muscle spasm	Use with opioids can be limited because of sedation
Anticonvulsants	Gabapentin Pregabalin	Neuropathic pain	Dose adjustment for renal insufficiency is required. Initiation requires close follow-up for titration
Anesthetics			
Local	Topical lidocaine 5% patch	Neuropathic pain	FDA approved to be worn for 12 h and then removed for 12 h
Systemic	Ketamine	Intractable pain	Dissociative anesthetic without significant respiratory depression Use of low-dose (sub-anesthetic) ketamine + opioids for treatment of acute cancer pain is an area of ongoing research
Intraoral	Magic/miracle mouthwash Lidocaine gel	Mucositis	Used in conjunction with parenteral opioids and good oral care regimens

 Table 56.3
 Adjuvant drugs for use during ED cancer pain crisis^a

^aOther adjuvant drugs include antidepressants, stimulants, bisphosphonates, and cannabinoids; these drugs are not included in the table because they are less likely to be used in the acute pain setting in the ED

antidepressants, corticosteroids, anticonvulsants, local anesthetics, muscle relaxants, and benzodiazepines. The addition of some of these medications may result in a better balance of patient safety and efficacy of pain relief. However, caution must be used to avoid oversedation or other side effects related to polypharmacy as many co-analgesics are potentially sedating.

Specific adjuvant medications that may be useful to emergency physicians for use in patients with pain crises are outlined in Table 56.3.

Psychosocial Support

There are many non-pharmacologic interventions that have been shown to be helpful for cancer pain, including physical interventions (e.g., massage, physical therapy), cognitivebehavioral interventions (e.g., breathing exercises, music), and spiritual interventions (e.g., chaplain support) [6]. While emergency physicians likely won't have access to all of these modalities when treating acute pain, it is important for them to provide psychosocial support and perform a focused assessment of the patient's support system, cognitive status, and level of psychological or spiritual stress as this will likely affect therapeutic outcomes [19, 45].

Opioid Safety

As referenced earlier, providers may feel tension when attempting to balance the use of opioids to achieve adequate analgesia with the current public health crisis of addiction, misuse, abuse, overdose, and death related to opioids. Cancer patients are also at risk for these same adverse outcomes associated with opioids. Using validated screening tools, one ED-based study found that one-third of cancer patients on opioids presenting to the ED were at high risk for opioid misuse and suggest screening for opioid misuse in the ED [46]. In addition, the majority of cancer patients (and parents of pediatric cancer patients) do not store opioids safely, save unused opioids, and are unaware of how to safely dispose of opioids [47, 48]. Finally, a recent study found that the incidence of ED visits for opioid overdose in patients with cancer doubled between 2006 and 2015 [49]. Despite the fact that cancer patients are at risk for adverse outcomes associated with opioids, emergency providers must still strive to effectively control pain. The principles listed below can help emergency providers treat cancer patients with uncontrolled pain in the ED while also maximizing safe opioid use [6, 34, 50].

- Use as low of a dose as possible to achieve adequate analgesia and minimize side effects
- Avoid combining opioids with other potentially sedating medications, particularly benzodiazepines
- Titrate doses carefully, particularly in patients at high risk of adverse effects (patients with decreased renal/hepatic function, lung disease, upper airway compromise, sleep apnea, and/or poor functional status)
- Review prescription drug monitoring program databases if available and monitor patients for signs of misuse/abuse of opioids.
- Work closely with primary care doctors, oncologists, and palliative care teams, relay any concerns regarding misuse/abuse of opioids, discuss any new/updated prescriptions, and ensure close follow-up
- Educate patients and caregivers about safe use, storage, and disposal of opioids
EPEC-EM

The Education for Physicians on End-of-Life Care (EPECTM) curriculum is intended to teach the core competencies of palliative care and is a useful resource for physicians wanting to learn more about this topic [51]. Supported by the National Institutes of Health, the EPEC-Emergency Medicine curriculum is an adaptation of the original EPEC curriculum designed for those who work in the emergency department. One of the tenets of care advocated by EPECTM-EM curriculum and others is the rapid titration of opioids to achieve pain control.

Rapid Titration

Rapid titration of opioids has been advocated to achieve adequate pain control for patients presenting to the ED rather than traditional outpatient oral medication titration which can require several days to achieve analgesia [52]. Several different protocols, utilizing various opioids, have been evaluated in the literature and found to be safe [53–55]. The commonalities between the recommended protocols include having a formal numerical assessment of pain followed by administration of medications based on severity of pain and a formal reassessment of pain at a scheduled interval (ranging from 5 to 30 min). These cycle or reassessment and medication administration continue until either pain control is achieved or unwanted side effects limit further opioid use.

EPEC-EM advocates the following approach to the rapid assessment of cancer pain [19, 34]:

Step 1: Assess the patient and history This assessment should include discussion of the home medication use and dosing of both opioids and adjuvants. Additionally, the responsiveness of the pain to opioids should be estimated.

Step 2: Administer treatment For patients with severe pain (>7/10), IV medications should be given. For those with mild to moderate pain, the best route and choice of medication can be individualized based on the assessment and goals of care.

- For opioid-naïve patients: 0.1 mg/kg of IV morphine equivalent (less if patient is elderly or high risk).
- For opioid-tolerant patients: 10% of the previous 24-h IV morphine equivalents.

These recommend starting doses are guidelines and should be customized to the clinical scenario and take into account the patients' presentation, polypharmacy, and goals of care.

Step 3: Reassess the patient's pain, and reassess for unwanted side effects (somnolence or confusion) when Cmax has been achieved

The timing of the reassessment can vary based on when the maximum concentration (C_{max}) of the medication is reached (calculated by route of administration, e.g., approximately 15 min for IV administration). This timing is chosen because the maximum side effects will be experienced at the time of C_{max} [19]. Although parental agents will reach C_{max} faster than orally or subcutaneously dosed opioids, there is some variation within parental agents as to the speed of achieving peak plasma concentrations. Fentanyl, which is more lipophilic than morphine, can achieve peak plasma concentrations within 5 min and therefore may be particularly useful for rapid titration [52, 55].

Step 4: Achieve adequate pain control by redosing the medication if necessary For patients with persistent severe pain (>7/10) in whom no unwanted side effects have been noted, the initial opioid dose can be doubled. For those in whom there has been an improvement in pain control, but an unacceptable level of pain persists, the initial dose can be repeated. Both of these strategies will increase the effective C_{max} and thereby decrease pain.

Steps 3 and 4 should be repeated until pain control is achieved or side effects limit further administration of opioids. Additionally, adjuvant medications should be considered.

Step 5: Determine the plan for disposition, discharge instructions, and follow-up Patients whose pain cannot be adequately controlled in the ED should be considered for admission. Similarly, patients may have other medical conditions aside from pain that prompted their ED visit and should be admitted if further evaluation or treatment is needed. In the case that a patient presented solely for pain and the rapid titration of pain medication in the ED has controlled their pain, discharge can be considered. Adjustments to the patients' long-acting and breakthrough opioids can be determined based on patients' previous medication use, allergies, and tolerance of medications in the past. Methadone should not be started or titrated in the ED without consultation because of its complicated dosing. Communication with the patients' treating primary care physician, oncologist, or palliative care specialist is important, as is ensuring follow-up for the discharged patient.

Palliative Sedation

Distinct from the rapid titration of medications, another strategy that can be considered in the ED for pain control in cancer patients with intractable pain is "palliative sedation." A commonly accepted definition for palliative sedation therapy is "the use of specific medications to relieve intolerable suffering from refractory symptoms by a reduction in patient consciousness" [56]. The goal of palliative sedation is to relieve distress and not to speed the dying process and therefore is distinct from euthanasia or physician-assisted suicide. Morphine has been used for the relief of dyspnea but is not effective at achieving sedation. Ketamine, a short-acting NMDA receptor antagonist, preserves respiration but can effectively achieve sedation and be used for intractable pain or agitation [57]. Emergency physicians will likely have some familiarity with ketamine from procedural sedations and use for acute, opioid-refractory pain, but when initiating it for palliative sedation, consultation with the palliative care team may be useful.

Consultation

Once the emergency physician has assessed the etiology of the cancer patient's pain, he or she may consider consulting specific services for assistance. Specific consulting services are outlined in Table 56.4.

Palliative Care in the Emergency Department

Consultation with the cancer patients' treating physicians, including their palliative care doctor, is useful not only in coordinating discharge but also in determining their treatment in the ED [58]. Some patients presenting to the ED may not have previously interacted with a palliative care physician; the emergency physician can and should consider consulting with palliative care if deemed necessary after an evaluation of the patient and discussion of wishes, regardless of patients' prior engagement with palliative care. The "Improving Palliative Care in Emergency Medicine" project has recommended the use of a screening tool to assist in the decision to consult palliative care from the ED [59]. If the

Table 56.4 Consulting services

Service	Indication
Oncologist	Discuss therapeutic plan, disposition
Palliative care	Symptom control, provide support
Pain specialists	Refractory symptoms, consideration for interventional strategies (e.g., nerve block, regional infusion)
Physical therapy, occupational therapy	Other modalities for pain control, improving pain related to movements
Psychiatry Social work Spiritual care	Relief of psychological pain Psychosocial support, relief of psychological pain Relief of suffering from unmet spiritual, existential needs

patient has a serious, life-threatening illness (in the case of cancer, this includes patients with metastatic or locally advanced incurable disease) and any one of the following conditions, palliative care consultation should be considered:

- Not surprised: You would not be surprised if the patient died in the next 12 months.
- Bounce backs: The patient has had >1 ED visit or hospital admission for the same condition within several months.
- Uncontrolled symptoms: The current ED visit is prompted by difficult-to-control physical or psychological symptoms.
- Functional decline: Their presentation reveals a decline in function, feeding intolerance, unintentional weight loss, or caregiver distress.
- Increasingly complicated: There exist complex long-term care needs that require more support.

The ED is an important point of contact for patients with palliative care needs, and, more recently, there has been growing interest in studying the effects of palliative care in the ED [60]. A recent study found that greater than 50% of patients with active cancer presenting to the ED had at least one unmet palliative care need (e.g., uncontrolled pain, feeling overwhelmed, difficulty with getting care/medications) [61]. Despite this apparent need, few palliative care consults are placed from the ED; one study showed that only 5% of consults to palliative care occurred in the ED [62]. While the ED is a fast-paced environment and there are perceived barriers to consultation, research shows that screening and referral for palliative care consultation are feasible in the ED [63, 64]. One study even found that they were able to effectively screen for unmet palliative care needs in the ED and then offer services to address those needs using existing ED resources (e.g., pharmacist and social worker) without increasing ED length of stay [61]. A 2020 systematic review of the existing literature found that palliative care in the ED is feasible, may improve quality of life, and does not appear to decrease survival; the authors, however, were unable to determine its effects on healthcare utilization [60]. Further research is needed to determine the most efficient and effective method of linking palliative care with emergency medicine.

Summary

Patients with cancer often seek care in the emergency department, for both pain related to their condition and for other symptoms. Regardless of the symptom prompting the visit, it is important to assess and manage pain in this population. Multiple classes of medication to acutely control pain are available to the emergency physician. It is important to assess the pain formally, using pain scales, and to discern if the pain crisis is related to progression of disease, expected breakthrough pain, or if a search for a new pain precipitant is needed. Regardless of the cause, pain should be managed quickly, and the patients' pain score should be frequently reassessed. The exact choice of medications will vary depending on the clinical situation and should be determined after reviewing the patients' home medications, comorbidities, and goals of care.

References

- van Den Beuken-van Everdingen MH, Hochstenbach LM, Joosten EA, Tjan-Heijnen VC, Janssen DJ. Update on prevalence of pain in patients with cancer: systematic review and meta-analysis. J Pain Symptom Manag. 2016;51(6):1070–90.
- Vandyk AD, Harrison MB, Macartney G, Ross-White A, Stacey D. Emergency department visits for symptoms experienced by oncology patients: a systematic review. Support Care Cancer. 2012;20(8):1589–99.
- Mayer DK, Travers D, Wyss A, Leak A, Waller A. Why do patients with cancer visit emergency departments? Results of a 2008 population study in North Carolina. J Clin Oncol. 2011;29(19):2683–8.
- Ahn S, Lee YS, Lim KS, Lee JL. Emergency department cancer unit and management of oncologic emergencies: experience in Asan Medical Center. Support Care Cancer. 2012;20(9):2205–10.
- Barbera L, Atzema C, Sutradhar R, Seow H, Howell D, Husain A, et al. Do patient-reported symptoms predict emergency department visits in cancer patients? A population-based analysis. Ann Emerg Med. 2013;61(4):427–37.e5.
- Swarm RA, Paice JA, Anghelescu DL, Are M, Bruce JY, Buga S, et al. Adult cancer pain. J Natl Compr Cancer Netw. 2019;17(8):977–1007.
- Caraceni A, Martini C, Zecca E, Portenoy RK, Ashby MA, Hawson G, et al. Breakthrough pain characteristics and syndromes in patients with cancer pain. An international survey. Palliat Med. 2004;18(3):177–83.
- Caraceni A, Davies A, Poulain P, Cortes-Funes H, Panchal SJ, Fanelli G. Guidelines for the management of breakthrough pain in patients with cancer. J Natl Compr Cancer Netw. 2013;11(Suppl 1):S29–36.
- American Pain Society. Principles of analgesic use in the treatment of acute pain and cancer pain. 6th ed. Glenview, IL: American Pain Society; 2008.
- Todd KH. Pain assessment instruments for use in the emergency department. Emerg Med Clin North Am. 2005;23(2):285–95.
- Brunelli C, Zecca E, Martini C, Campa T, Fagnoni E, Bagnasco M, et al. Comparison of numerical and verbal rating scales to measure pain exacerbations in patients with chronic cancer pain. Health Qual Life Outcomes. 2010;8:42.
- 12. Hjermstad MJ, Fayers PM, Haugen DF, Caraceni A, Hanks GW, Loge JH, et al. Studies comparing numerical rating scales, verbal rating scales, and visual analogue scales for assessment of pain intensity in adults: a systematic literature review. J Pain Symptom Manag. 2011;41(6):1073–93.
- Dalal S, Hui D, Nguyen L, Chacko R, Scott C, Roberts L, et al. Achievement of personalized pain goal in cancer patients referred to a supportive care clinic at a comprehensive cancer center. Cancer. 2012;118(15):3869–77.

- Knudsen AK, Aass N, Fainsinger R, Caraceni A, Klepstad P, Jordhoy M, et al. Classification of pain in cancer patients—a systematic literature review. Palliat Med. 2009;23(4):295–308.
- Pachman DR, Barton DL, Swetz KM, Loprinzi CL. Troublesome symptoms in cancer survivors: fatigue, insomnia, neuropathy, and pain. J Clin Oncol. 2012;30:3687–96.
- Barbera L, Sutradhar R, Howell D, Corn E, O'Brien MA, Seow H, et al. Factors associated with opioid use in long-term cancer survivors. J Pain Symptom Manag. 2019;58(1):100–7.
- Greco MT, Roberto A, Corli O, Deandrea S, Bandieri E, Cavuto S, Apolone G. Quality of cancer pain management: an update of a systematic review of undertreatment of patients with cancer. J Clin Oncol. 2014;32(36):4149–54.
- Patel PM, Goodman LF, Knepel SA, Miller CC, Azimi A, Phillips G, Gustin JL, Hartman A. Evaluation of emergency department management of opioid-tolerant cancer patients with acute pain. J Pain Symptom Manag. 2017;54(4):501–7.
- Desandre PL, Quest TE. Management of cancer-related pain. Emerg Med Clin North Am. 2009;27(2):179–94.
- Meghani SH, Byun E, Gallagher RM. Time to take stock: a metaanalysis and systematic review of analgesic treatment disparities for pain in the United States. Pain Med. 2012;13(2):150–74.
- Platts-Mills TF, Esserman DA, Brown DL, Bortsov AV, Sloane PD, McLean SA. Older US emergency department patients are less likely to receive pain medication than younger patients: results from a national survey. Ann Emerg Med. 2012;60(2):199–206.
- 22. Todd KH, Ducharme J, Choiniere M, Crandall CS, Fosnocht DE, Homel P, et al. Pain in the emergency department: results of the pain and emergency medicine initiative (PEMI) multicenter study. J Pain. 2007;8(6):460–6.
- 23. Lee P, Le Saux M, Siegel R, Goyal M, Chen C, Ma Y, Meltzer AC. Racial and ethnic disparities in the management of acute pain in US emergency departments: meta-analysis and systematic review. Am J Emerg Med. 2019;37(9):1770–7.
- Centers for Disease Control and Prevention (CDC). Vital signs: overdoses of prescription opioid pain relievers—United States, 1999– 2008. MMWR Morb Mortal Wkly Rep. 2011;60(43):1487–92.
- 25. Centers for Disease Control and Prevention (CDC). Annual Surveillance Report of Drug-Related Risks and Outcomes — United States. Surveillance Special Report. Centers for Disease Control and Prevention, U.S. Department of Health and Human Services. Published 31 Aug 2018.
- 26. Barbera L, Sutradhar R, Chu A, Seow H, Howell D, Earle CC, et al. Opioid prescribing among cancer and non-cancer patients: time trend analysis in the elderly using administrative data. J Pain Symptom Manag. 2017;54(4):484–92.
- Portenoy RK. Treatment of cancer pain. Lancet. 2011;377(9784):2236–47.
- Portenoy RK, Ahmed E. Principles of opioid use in cancer pain. J Clin Oncol. 2014;32(16):1662–70.
- World Health Organization. Cancer pain relief. 2nd ed. Geneva, Switzerland: World Health Organization; 1996.
- Raffa RB, Pergolizzi JV Jr. A modern analgesics pain 'pyramid'. J Clin Pharm Ther. 2014;39(1):4–6.
- Vargas-Schaffer G. Is the WHO analgesic ladder still valid? Twentyfour years of experience. Can Fam Physician. 2010;56(6):514–7. e202–5.
- Pathan H, Williams J. Basic opioid pharmacology: an update. Br J Pain. 2012;6(11):11–6.
- Wiffen PJ, Derry S, Moore RA. Tramadol with or without paracetamol (acetaminophen) for cancer pain. Cochrane Database Syst Rev. 2017;5(5):CD012508.
- Emanuel LL, Quest T, editors. "Module 12: malignant pain." The Education in Palliative and End-of-life Care (EPEC) for Emergency Medicine Curriculum: © EPEC; 2010.

- Moryl N, Coyle N, Foley KM. Managing an acute pain crisis in a patient with advanced cancer: "this is as much of a crisis as a code". JAMA. 2008;299(12):1457–67.
- Fine PG, Portenoy RK. Establishing "best practices" for opioid rotation: conclusions of an expert panel. J Pain Symptom Manag. 2009;38(3):418–25.
- Knotkova H, Fine PG, Portenoy RK. Opioid rotation: the science and the limitations of the equianalgesic dose table. J Pain Symptom Manag. 2009;38(3):426–39.
- Cherny N, Ripamonti C, Pereira J, Davis C, Fallon M, McQuay H, et al. Strategies to manage the adverse effects of oral morphine: an evidence-based report. J Clin Oncol. 2001;19(9):2542–54.
- McNicol E, Horowicz-Mehler N, Fisk RA, Bennett K, Gialeli-Goudas M, Chew PW, et al. Management of opioid side effects in cancer-related and chronic noncancer pain: a systematic review. J Pain. 2003;4(5):231–56.
- 40. Librach SL, Bouvette M, De Angelis C, Farley J, Oneschuk D, Pereira JL, et al. Consensus recommendations for the management of constipation in patients with advanced, progressive illness. J Pain Symptom Manag. 2010;40(5):761–73.
- Lee-Robichaud H, Thomas K, Morgan J, Nelson RL. Lactulose versus polyethylene glycol for chronic constipation. Cochrane Database Syst Rev. 2010;7:CD007570.
- Thomas J, Karver S, Cooney GA, Chamberlain BH, Watt CK, Slatkin NE, et al. Methylnaltrexone for opioid-induced constipation in advanced illness. N Engl J Med. 2008;358(22):2332–43.
- Chey WD, Webster L, Sostek M, Lappalainen J, Barker PN, Tack J. Naloxegol for opioid-induced constipation in patients with noncancer pain. N Engl J Med. 2014;370(25):2387–96.
- 44. Lussier D, Huskey AG, Portenoy RK. Adjuvant analgesics in cancer pain management. Oncologist. 2004;9(5):571–91.
- 45. Albiach CF, Estevez FV, Alarcon MDL, de Madariaga M, Carregal A, Arranz J, et al. Real-life management of patients with break-through cancer pain caused by bone metastases in Spain. J Pain Res. 2019;12:2125–35.
- Reyes-Gibby CC, Anderson KO, Todd KH. Risk for opioid misuse among emergency department cancer patients. Acad Emerg Med. 2016;23(2):151–8.
- 47. Madden K, Reddy AS, De La Cruz MJ, Liu DD, Bruera E. Patterns of storage, use, and disposal of prescription opioids by parents of children with cancer. J Pain Symptom Manag. 2020;59(2):320–6.
- 48. Silvestre J, Reddy A, de la Cruz M, Wu J, Liu D, Bruera E, Todd KH. Frequency of unsafe storage, use, and disposal practices of opioids among cancer patients presenting to the emergency department. Palliat Support Care. 2017;15(6):638–43.
- Jairam V, Yang DX, Yu JB, Park HS. Emergency department visits for opioid overdoses among patients with cancer. J Natl Cancer Inst. 2020;112(9):938–43.
- 50. Centers for Disease Control and Prevention. CDC Guideline for Prescribing Opioids for Chronic Pain [Internet]. Centers for

Disease Control and Prevention, U.S. Department of Health and Human Services; 2019 [cited 11 Feb 2020]. https://www.cdc.gov/ drugoverdose/prescribing/guideline.html.

- Robinson K, Sutton S, von Gunten CF, Ferris FD, Molodyko N, Martinez J, et al. Assessment of the education for Physicians on End-of-Life Care (EPEC) Project. J Palliat Med. 2004;7(5):637–45.
- Mercadante S. Opioid titration in cancer pain: a critical review. Eur J Pain. 2007;11(8):823–30.
- 53. Chang AK, Bijur PE, Campbell CM, Murphy MK, Gallagher EJ. Safety and efficacy of rapid titration using 1 mg doses of intravenous hydromorphone in emergency department patients with acute severe pain: the "1+1" protocol. Ann Emerg Med. 2009;54(2):221–5.
- Hagen NA, Elwood T, Ernst S. Cancer pain emergencies: a protocol for management. J Pain Symptom Manag. 1997;14(1):45–50.
- 55. Soares LG, Martins M, Uchoa R. Intravenous fentanyl for cancer pain: a "fast titration" protocol for the emergency room. J Pain Symptom Manag. 2003;26(3):876–81.
- 56. de Graeff A, Dean M. Palliative sedation therapy in the last weeks of life: a literature review and recommendations for standards. J Palliat Med. 2007;10(1):67–85.
- Shlamovitz GZ, Elsayem A, Todd KH. Ketamine for palliative sedation in the emergency department. J Emerg Med. 2013;44(2):355–7.
- Mierendorf SM, Gidvani V. Palliative care in the emergency department. Perm J. 2014;18(2):77–85.
- 59. Lamba S, DeSandre PL, Todd KH, Bryant EN, Chan GK, Grudzen CR, et al. Integration of palliative care into emergency medicine: the Improving Palliative Care in Emergency Medicine (IPAL-EM) collaboration. J Emerg Med. 2014;46(2):264–70.
- 60. Wilson JG, English DP, Owyang CG, Chimelski EA, Grudzen CR, Wong HN, Aslakson RA. End-of-life care, palliative care consultation, and palliative care referral in the emergency department: a systematic review. J Pain Symptom Manag. 2020;59(2):37283.
- Reuter Q, Marshall A, Zaidi H, Sista P, Powell ES, McCarthy DM, Dresden SM. Emergency department-based palliative interventions: a novel approach to palliative care in the emergency department. J Palliat Med. 2019;22(6):649–55.
- 62. Grudzen CR, Hwang U, Cohen JA, Fischman M, Morrison RS. Characteristics of emergency department patients who receive a palliative care consultation. J Palliat Med. 2012;15(4):396–9.
- 63. Grudzen CR, Richardson LD, Hopper SS, Ortiz JM, Whang C, Morrison RS. Does palliative care have a future in the emergency department? Discussions with attending emergency physicians. J Pain Symptom Manag. 2012;43(1):1–9.
- 64. George N, Phillips E, Zaurova M, Song C, Lamba S, Grudzen C. Palliative care screening and assessment in the emergency department: a systematic review. J Pain Symptom Manag. 2016;51(1):108–19.

e-mail: Ashley.shreves@ochsner.org

Ochsner Medical Center, New Orleans, LA, USA

Palliative Medicine & Supportive Care, Emergency Medicine,

A. Shreves (🖂)

Dyspnea in the Dying Patient

Ashley Shreves

Case Study

A 65-year-old female presents with severe shortness of breath. On review of the medical record, this patient has widely metastatic lung cancer, diagnosed 3 years prior, and recent imaging studies have shown progression of disease, despite chemotherapy and radiation. In the past 2 weeks, she has become weaker and has lost all interest in food. On review of the oncologic clinic notes, there has been discussion about enrolling the patient in a clinical trial, but her oncologist also suggested that she consider enrolling in hospice. In the ED, the patient's triage VS are abnormal: HR 120, RR 35, O2 sat 80%, BP 110/60, and temp 100. The patient is too ill to engage in discussions about goals of care, and her husband, who is her healthcare power of attorney, has not arrived to the hospital yet. The nurse asks: "Should we get everything ready for you to intubate the patient?" As there are no documented advance directives in the patient's electronic medical record and no surrogate decision-maker available for discussion, you decide to give patient a trial of noninvasive ventilation, while ordering labs and a chest X-ray, with the hopes of identifying the underlying etiology of her dyspnea. The patient initially tolerates the BIPAP well, with her RR improving from 35 to 25 and her oxygenation improving from 80 to 95%. She also appears more comfortable. Her labs show multiple abnormalities, including acute renal failure, an elevated troponin, a lactate of 5, and a WBC count of 20. Chest X-ray shows a small effusion and diffuse pulmonary metastatic disease, increased from the prior imaging study. When the patient's husband arrives, he tells you that his wife had actually decided to enroll in hospice but they hadn't had a chance yet and they panicked this morning when she became short of breath. He's certain that she would want to go home today, if hospice care could be arranged. You administer 2 mg morphine intravenously and

this further improves her dyspnea. Thirty minutes later, she needs an additional dose: this time 4 mg is given. She is then weaned from the BIPAP and is comfortable on 4 L of oxygen, delivered via nasal cannula. Hospice is called and says they can manage the patient's dyspnea at home and can have all of the necessary equipment available, including oxygen, by the afternoon. Both the patient and her husband are comfortable with this plan, and the patient is discharged home with hospice care from the ED.

Introduction

Dyspnea is a broad, general term used to characterize any sensation of respiratory discomfort. In a statement by the American Thoracic Society from 2012, dyspnea was appropriately defined as "a subjective experience of breathing discomfort that consists of qualitatively distinct sensations that vary in intensity" [1]. It is important to emphasize that dyspnea is indeed subjective and thus a symptom; this contrasts to classic signs of respiratory distress including tachypnea, nasal flaring, and accessory muscle activation. As such, a variety of language may be used to describe the sensation: terms such as breathlessness, shortness of breath, tightness, air hunger, difficulty breathing, labored breathing, and heavy breathing. All of these descriptors relate to an increased awareness of the breathing process-normally an unconscious physiologic activity-caused by any number of insults to be described later in this chapter.

As a common endpoint for multiple disease processes, dyspnea is remarkably prevalent in the advanced cancer population at the end of life. Some degree of dyspnea has been reported in up to one-third of all older adults living at home, approximately half of all patients admitted to tertiary care hospitals, 70–80% of patients with terminal cancer in the last 6 weeks of life, and up to 94% of patients with chronic lung disease at some point in the last year of life [2–6]. It is both debilitating for patients and emotionally upsetting for their families and caregivers. It also ranks among the most



[©] Springer Nature Switzerland AG 2021

K. H. Todd et al. (eds.), Oncologic Emergency Medicine, https://doi.org/10.1007/978-3-030-67123-5_57

distressing symptoms at the end of life, leading to a marked reduction in the quality of life and a source of both fear and anxiety for all parties involved [7].

Derangements of the pulmonary/respiratory system have long been recognized by emergency providers as an indicator of serious illness and of the potential need for prompt intervention. Patients who present to the emergency department (ED) in respiratory distress appear extremely ill and receive rapid attention, in many cases leading to intubation and the use of mechanical ventilation. When caring for a patient with end-of-life/terminal dyspnea, however, for whom these aggressive interventions may be misaligned with patient's goals of care, the necessity for rapid treatment is no less imperative. In these situations, a focused and patient-centric plan coupled with an empiric approach to symptom management is necessary. This chapter will present a structured approach to the management of dyspneic patient at the end of life whose goals are not purely curative and for whom comfort and quality of life are of primary importance. As emergency providers are well aware, responding to acute symptomatology often requires treatment prior to definitive diagnosis, and the management of terminal dyspnea is no different. In some instances, however-namely, those with reversible causes-diagnosis is worth pursuing, and those instances will be discussed herein.

Neurophysiology

The pathophysiology of dyspnea is complex and remains poorly understood compared to other common clinical phenomena such as pain or nausea. The best current evidence counters the traditional model of dyspnea as a singular entity, instead suggesting multiple neurophysiologically distinct afferent pathways each replete with a unique subjective experience and a different set of predisposing stimuli [8, 9]. To illustrate, one such multifaceted model subdivides dyspnea into (1) air hunger, which is the urge to breathe characterized by an increased ventilatory drive secondary to hypercapnia; (2) increased work of breathing, associated with exhaustion due to effortful breathing; and (3) tightness, a sensation most associated with bronchoconstriction and asthma [10]. While the details of this particular multidimensional model of dyspnea are beyond the necessary scope of knowledge for the emergency provider, it is worth noting that data will continue to emerge on these discrete neurologic pathways, with potential implications for specifically targeted therapeutic options in the future. It also highlights the need for careful appraisal of the literature concerning dyspnea and potential for conflicting study outcomes, as the majority of evidence to date is based on the assumption of a single unified perception of dyspnea.

A more practical understanding of dyspnea as it relates to oncologic etiologies focuses primarily on the brainstem respiratory complex and its relation to the most common variant of dyspnea, air hunger [11]. Through direct action on medullary chemoreceptors, PaCO₂, PaO₂, and pH act to regulate respiratory drive by modulating both rate and effort [12]. When an insult causes disruption in homeostasis of any of these variables, an appropriate motor response is elicited to correct this imbalance. The end result of these physiologic insults, whether they be mechanical, metabolic, or neuromuscular, is often a failure to match ventilation with brainstem-mediated respiratory drive. Once this mismatch occurs, the patient develops a sensation of breathlessness and air hunger, which may then exacerbate the predisposing condition through increased metabolic demand. Anxiety and fear, sensations which both precipitate and are exacerbated by dyspnea, are cortically mediated (primarily limbic and paralimbic) and are distinct from those initial sensations mediated by the brainstem respiratory motor drive [13]. This neurologic distinction allows for multiple therapeutic approaches to be discussed.

Etiology and Prevalence

Dyspnea is widely prevalent in advanced cancer, with a marked increase in symptoms as patients near the end of life [4]. Furthermore, dyspnea is both commonly present in advanced cancer patients presenting to the ED and a predictor of death in this population [14]. While primary lung cancer conveys the highest risk of all malignancies for developing shortness of breath—affecting up to 84% in one large crosssectional study—all primary cancers are associated with some increased incidence of dyspnea [3, 15]. There are many specific conditions associated with respiratory compromise which disproportionately affect patients with active malignancies. Below is a partial list of these diagnoses with particular emphasis on cancer-related etiologies and common comorbid conditions. Many are reversible, and these will be explored individually later in this chapter [16].

- Airway obstruction: foreign body/aspiration and tumor burden
- Anaphylaxis and angioedema
- Asthma/reactive airway disease
- Anemia: blood loss, nutritional deficiency, and chemotherapyinduced
- · Behavioral/emotional: anxiety and panic attack
- Chronic obstructive pulmonary disease/emphysema
- Deconditioning/cachexia/muscle weakness
- Decreased cardiac output: myocardial ischemia, arrhythmia, tamponade, and hypovolemia

- Decreased chest wall compliance: hepatosplenomegaly, ascites, obesity, and tumor burden
- Diaphragmatic/respiratory muscle weakness: neuromuscular disorders and fatigue
- Metabolic acidosis: renal failure, sepsis, and toxic ingestions
- Pneumonia
- Pneumothorax
- Pulmonary edema/congestive heart failure
- Pulmonary effusion: malignant and infectious
- Pulmonary embolism
- Pulmonary hypertension
- Pulmonary fibrosis/interstitial lung disease: autoimmune, environmental, and secondary to radiation or chemotherapy
- · Pulmonary receptor stimulation: environmental irritants
- Ventilation/perfusion mismatch

Complicating diagnosis and treatment, most patients will present with multiple, coexisting etiologies of dyspnea; a study of 100 advanced cancer patients revealed a median of 5 different abnormalities that could have contributed to their shortness of breath [17]. The most frequent cause for symptoms in this cohort was a direct pulmonary pathology related to disease progression; this was followed by treatmentrelated pathologies secondary to chemotherapy or radiation and then by nonmalignant etiologies such as underlying chronic lung disease. There were, in up to 30% of patients with dyspnea and advanced cancer, no clear identifiable causes for their dyspnea [17].

Evaluation

Arrival in ED/History

Evaluation of the undifferentiated patient with respiratory distress may be challenging in the emergency setting. Frequently, providers face an acutely unstable patient with limited access to complete medical history. Patients may arrive initially unaccompanied by family or caregivers to provide historical context to the hospital visit. In these situations, management should proceed as with any other emergency patient: with focus on stabilization of the airway, breathing, and circulation, as represented through legal advance directives. In many states, the increasingly utilized POLST (Physician Orders for Life-Sustaining Interventions) advanced directive contains the only legally recognized outof-hospital Do Not Intubate (DNI) order and can provide critical guidance in the early management of unstable patients with advanced cancer. If the situation arises in which an intervention was made-such as endotracheal intuba-

Physical Exam

There are many elements of the physical exam in the dyspneic cancer patient which can aid in both prognostication and diagnosis. As patients may be unable to communicate either secondary to their chronic disease process or from their dyspnea, exam findings must be used to guide acute medical management. Classic signs associated with respiratory distress include gasping, accessory muscle activation, tachypnea, shallow respirations, and poor air movement on lung auscultation. Facial expressions should also be noted, as grimacing may suggest pain or discomfort from dyspnea [16]. Family members or caregivers may be especially helpful in this regard, as they may be more skilled at distinguishing specific nonverbal cues or changes from baseline appearance [19]. Chest auscultation for abnormal breath sounds may also guide management-diffuse wheezing may suggest an obstructive process such as asthma/COPD, while bibasilar or focal rales may be more indicative of pulmonary edema or pneumonia. Stridor is concerning for upper airway obstruction or allergic reaction. A focal or asymmetric decrease in breath sounds could indicate pulmonary effusion, pneumothorax, or hemothorax. Distant cardiac sounds or a cardiac rub is concerning for pericardial effusion. In patients with notable skin pallor, anemia may be contributing to dyspnea, while peripheral cyanosis would suggest a more significant hypoxic state. Abdominal distention may represent underlying malignant ascites, which may cause restriction in lung expansion and subsequent respiratory distress. Peripheral edema, especially a change from baseline, may indicate worsening heart or renal failure, while asymmetric extremity edema may be concerning for deep vein thrombosis and pulmonary embolism.

There is prognostic value in certain physical exam findings at the end of life. For example, the inability to clear oral secretions, colloquially known as a "death rattle," has been associated with a median time from onset to death of 23 h [20]. Other physical exam findings to guide expectations include respirations with mandibular movements (median time 2.5 h), extremity cyanosis (1 h), and inability to palpate radial pulse (1 h) [20]. A retrospective chart review of advanced cancer patients revealed the following historical and vital sign abnormalities as predictors of death within 2 weeks: triage respirations >28 (RR 12.7), pulse >110 (RR 4.9), history of uncontrolled progressive disease despite treatment (RR 21.9), and history of metastatic disease (RR 3.9) [21].

Laboratory Studies

There is no strong evidence to guide the decision to send laboratory studies in the acute emergency setting for the terminal cancer population. Consider sending labs if they could provide diagnostic clues which will lead to action consistent with the patient's stated goals. In some patients, the placement of an intravenous line alone is an unwelcome burden.

If mechanical ventilation is an option, assessment of $PaCO_2$ may be of clinical utility. If the patient wishes to receive blood transfusions, a complete blood count should be checked, along with a type and screen. While lactate levels have been shown to correlate with mortality and thus could theoretically be used as a prognostic tool, a study on patients with advanced cancer found that arterial blood gas could not help differentiate between patients who died imminently and those who did not [21, 22].

EKG

An electrocardiogram is a noninvasive intervention which provides rich diagnostic information and minimal patient burden. An EKG can provide diagnostic clues to indicate myocardial infarction, dysrhythmia, pericarditis/myocarditis, pericardial effusion, pulmonary embolism/heart strain, electrolyte abnormalities, or digoxin effect. In the absence of any compelling reason not to, or if refused by the patient, obtaining a 12-lead EKG should be part of the ED evaluation of the dying patient with dyspnea.

Imaging

Chest X-ray is similar to the electrocardiogram in terms of great diagnostic value compared to minimal burden. Common etiologies of dyspnea may be diagnosed rapidly in the emergency setting with chest radiography: pleural effusions, pneumothoraces, and pulmonary infiltrates/edema are all easily identifiable and may allow for target therapies to relieve symptoms. For patients with limited functional status, portable films may be shot at the bedside. As an adjunct, or even a potential replacement, for chest radiography is bedside ultrasound. Emergency physicians are becoming more adept at using ultrasound as a diagnostic tool and as a procedural aide, and the noninvasive nature of the modality allows for highquality images to be collected with a minimum burden to the patient. Recent studies have shown that thoracic ultrasound can differentiate between cardiac and pulmonary causes of dyspnea and accurately diagnose free or loculated pleural effusions, pneumothoraces, and lung consolidations [23, 24].

Generally, there is limited value in computed tomography (CT) for the diagnosis and treatment of dyspnea in terminal

cancer patients. While CT represents the gold standard for detection of pulmonary embolus (PE) and therefore should be used if the patient's goals of care and functional status align with PE treatment protocols, this test requires that the patient briefly leave their monitored bed and move to a radiology suite where close symptom management is challenging and onto a flat exam table which may exacerbate dyspnea [25]. Also, since it is reasonable to discontinue or hold anticoagulation for treatment of venous thromboembolism when advanced cancer patients enter the dying phase, it extends that withholding anticoagulation for acute PE is also reasonable; these patients are unlikely to see significant long-term benefit from anticoagulation and are at higher risk for complications [26]. Symptomatic management can and should continue despite the lack of a concrete diagnosis, so if PE is suspected, the decision to pursue advanced imaging should come only after a frank discussion with patient and family about the risks of harm and benefit of the proposed treatment course.

Cardiac Monitoring/Telemetry/Vital Signs

Cardiac monitoring provides real-time information to providers and can also help to identify transient dysrhythmias. However, it also provides a noisy and oftentimes fearproducing distraction for patients and their families. Unless there is a clear and convincing reason to keep patients on cardiac monitoring, consider changing alarm limits, turning off in-room monitor screens, or removing the patient entirely to minimize physical barriers between patient and family. Vital sign abnormalities should be expected in the dying patient, and unless the rapid identification of these vital signs will make a meaningful impact in patient care, it may be best to keep them off. This will allow the family to focus on their loved one without distractions and remind providers that treatment of the patient should come before treatment of vital sign abnormalities in this population.

General Management

There exist two approaches to the alleviation of dyspnea, as there are with most acute symptoms: One approach is to correct the underlying disorder responsible for the insult in physiologic hemostasis. For example, this may include the drainage of a symptomatic pleural effusion or pericardial effusion. It may involve provision of supplemental inhaled oxygen to a hypoxic patient or bronchodilators to an asthmatic. In the acute setting, however, diagnostic uncertainty exists, and a direct approach is not always possible. And in some scenarios, the medical treatment necessary for correction of a primary insult may come with an unacceptable side effect profile or burden to the patient being treated. In these situations, a second approach is necessary to alleviate dyspnea by interfering with the downstream cortical pathway. This is not a departure from standard emergency practice many patients present to the ED with undifferentiated pain or nausea, which requires prompt symptomatic treatment prior to availability of diagnostic testing.

Opioids

The general approach to nonspecific, terminal dyspnea primarily consists of systemic opioid administration. Opioids are safe, effective, and largely predictable and fall well within the comfort zone of the emergency provider. The majority of laboratory and clinical trials to date suggest a benefit of opioids for the treatment of symptomatic breathlessness in advanced illness [11, 27]. But similar to the pathophysiology of dyspnea itself, the exact mechanisms by which opioids exert their influence and alleviate dyspnea are not entirely understood. Leading theories based on current experimental evidence indicate that opioids likely function to modulate the effect of chemoreceptor-activated central respiratory drive on actual ventilation rate and effort [28]. By reducing this reaction to insult and the subsequent compensatory physiologic changes, an increased subjective tolerance may be reached. In addition, the established presence of opioid receptors in bronchial epithelial cells indicates a potential function in both central feedback and local inflammatory response [29, 30]. Finally, there exists a strong emotional component of anxiety which is commonly reported alongside dyspnea and acts to exacerbate the subjective experience. Though it is unclear whether there exists a direct or indirect effect, opioid administration has been shown to measurably decrease reported anxiety in dyspneic patients [30].

In 2011, Banzett and colleagues performed a wellcontrolled randomized trial using morphine for the relief of dyspnea [31]. This study artificially stimulated air hunger by limiting minute ventilation in healthy patients while inducing hypercapnia. Using patient-reported dyspnea scores on the validated VAS as their primary outcome, IV morphine was compared to IV saline with a significant benefit in both dyspnea scores and anxiety in the morphine study arm. While this study was performed on young and healthy subjects and therefore is less representative of the typical cancer patient, it provides insight into the mechanics of dyspnea and offers valuable data from a controlled environment.

Larger analyses of clinical data have also revealed similar findings. In a 2015 systematic review, opioids demonstrated modest effectiveness for dyspnea in cancer patients. Most of the included studies evaluated morphine. The quality of the included studies varied, with only 4 of the 14 including a placebo arm [32].

The choice of opioid medication for the relief of breathlessness is based on provider preference and departmental availability, similar to the treatment of pain. The literature bears this to be true, as dyspnea studies often use different, but generally equivalent, opioid regimens. There have been clinically significant results shown in trials with oral dihydrocodeine [33], oral hydromorphone [34], IV morphine [35], and oral morphine [36] and multiple studies on subcutaneous morphine [37]. Renal dysfunction has been cited as a justification for caution in morphine administration secondary to the theoretical risk of limited renal clearance of toxic metabolites, although the data for this is not robust [38]. Fentanyl has no clinically significant active toxic metabolites and may be effectively used if there is provider concern [39].

Route of administration should be based on patientspecific parameters; in patients with intravenous access, for whom peripheral IV placement is within their established goals of care and does not represent and unwelcome burden, IV administration is straightforward and rapid. These medications can also be delivered subcutaneously in patients without IV access with minimal discomfort and may also be given orally for those patients able to safely swallow. As was described earlier, there is no strong data to support the use of nebulized opioid formulations, despite the potential for benefit given the known presence of local opioid receptors within lung epithelial cells [30]. However, it has been suggested that many of these nebulized trials have failed to show a difference against placebo due to the fact that nebulized saline could also be an effective treatment modality [27]. Finally, there have been no randomized controlled trials on formulations other than morphine for nebulized delivery, so additional investigation must be performed prior to ruling out nebulized opioids altogether [37].

In terms of dosing and escalation, providers should treat opioid administration for dyspnea similar to pain, by giving a reasonable starting dose and reassessing symptom progression in 10-15-min intervals, up-titrating as needed [16]. Recent prospective trials have used mean doses of 2.5 mg of PO hydromorphone (equivalent to 0.4 mg IV) and 9.4 ± 8.8 mg of PO morphine (equivalent to 0.8 mg IV) to achieve a desired level of patient comfort. Reasonable starting doses for opioid-naïve patients, therefore, should be approximately 1-2 mg IV morphine equivalent or 0.2-0.4 mg IV hydromorphone. If symptoms are unchanged after 10-15 min, consider re-dosing an equivalent or increased amount. If symptoms are improved, but not fully controlled, consider giving another 50% of the starting dose, and continue to reassess. Be aware that higher doses may be necessary in patients who are opioid tolerant; many advanced cancer patients will be on chronic standing and/or breakthrough opioids for chronic pain. In these situations, start by administering approximately 10% of the patient's total daily

opioid dose. For example, if a patient takes 15 mg of oral morphine every 4 h around the clock, their total daily dose equals 90 mg. An appropriate initial dose would be 9 mg PO morphine (equivalent to 3 mg IV). If the patient is taking multiple formulations of opioids, a conversion table should be employed to ensure safe dosing practices. Complicated dosing or high-dose opioid regimens should warrant a consultation with the palliative care service, if available.

Measure improvement in dyspnea through direct patient report, if possible. Again, similar to pain management, the patient's subjective experience of their symptoms is the best indicator of improvement. A visual analog scale (VAS) is usually used for these purposes in both research and clinical use. However, as patients near the end of life are oftentimes unable to effectively communicate, providers will often be forced to rely on elements of the physical exam and family gestalt to gauge response to treatment. It has been suggested to use simple "yes/no" questioning for symptom relief in patients struggling to communicate, as many patients who are unable to provide a scaled response to symptom improvement are still able to indicate yes/no responses [19]. There has also been a respiratory distress observation scale developed for these clinical scenarios, which involves heart rate, respiratory rate, degree of restlessness, accessory muscle use, end-expiratory grunting, nasal flaring, and a "look of fear" [40]. These tools are necessary to prevent undertreatment of a patient's symptoms due to communication challenges.

It is important to consider the potential side effects associated with opioid administration, chief among them being constipation. For this reason, any patient placed on scheduled opioid treatment should be managed expectantly with a bowel regimen. Common prophylactic regimens include a stool softener (e.g., docusate, polyethylene glycol, magnesium) with a stimulant laxative (e.g., senna, bisacodyl) [41]. Other side effects to consider include nausea, vomiting, sedation, pruritus, and allergy/anaphylaxis. These can be managed supportively.

There is understandable hesitation on the part of emergency physicians when considering the use of opioids in patients with dyspnea. In sufficient doses, opioids can lead to respiratory depression and even apnea, so administering these medications to a patient with respiratory compromise can seem dangerous. Fortunately, there is evidence to support the safety of these agents in cancer populations suffering from dyspnea, particularly those patients at the EOL. In several small prospective observational studies, mostly comprised of patients with advanced cancers, carefully administered opioids, even when given to opioid-naïve patients, did not result in clinically important respiratory depression or hypercapnia [42–44]. Furthermore, two large observational studies of hospice patients found minimal to no association with opioid usage, dosage, and life expectancy [45]. The key to safe, effective opioid administration, as stated earlier, is to "start low, go slow" carefully targeting symptomatic relief, titrating up doses in appropriate intervals and with appropriate clinical monitoring.

Benzodiazepines

In select clinical scenarios, administration of benzodiazepines may be considered as a second- or third-line pharmacologic agent. The primary indication for this medication is when anxiety appears to be playing a significant role in the patient's discomfort and when other interventions have failed to alleviate symptoms. As briefly stated earlier, there is a physiologic justification for benzodiazepines despite the current lack of strong supporting clinical evidence. In current neurophysiologic models of dyspnea, there appears to be a distinct emotional component of dyspnea, which may be modulated by opioids but may theoretically benefit more from the anxiolysis associated with benzodiazepine administration [10].

There is, however, a very weak clinical evidence base for this intervention with few studies indicating effectiveness. The largest review to date published in 2016 examined 8 independent trials composed of 214 subjects and found no benefit to these agents when compared to placebo or opioids [46]. Compared to placebo, somnolence and drowsiness were observed more frequently in patients receiving benzodiazepines. Surprisingly, when midazolam and morphine were directly compared, morphine was associated with more adverse events.

Given limited evidence for efficacy but without evidence for serious harm when used in small doses, benzodiazepines should be considered as an adjunct to opioids in situations where anxiety appears to be playing a role in symptomatology [47]. A starting dose of lorazepam 1 mg (PO or parenteral) or diazepam 5 mg (PO or parenteral) would be appropriate in most patients.

Corticosteroids

Corticosteroids represent a treatment for certain conditions associated with advanced cancers and should not be applied generally in the same fashion as opioids and benzodiazepines. There is limited data on their use, but case studies report temporary symptomatic improvement for specific conditions such as lymphangitic malignant spread and chemotherapy-/ radiation-induced pneumonitis [48]. Corticosteroids are the mainstay of treatment for radiation-induced lung injury, typically treated with a taper starting with 60–100 mg of daily oral prednisone [49]. There are also case reports of rapid improvement in dyspnea associated with upper airway tumor obstruction after administration of steroids, likely secondary to reduction in airway edema [50]. Reported doses include 10 mg IV dexamethasone and 125 mg IV methylprednisolone, administered every 6 h. A recent systematic review found no benefit to steroids in cancer patients with dyspnea, though the quality of the included studies was low, limiting the ability to draw firm conclusions [51]. Significant side effects exist and must be considered. These include hyperglycemia, infection risk, fluid retention, and potential psychomotor agitation [52]. In patients with life expectancies of days to weeks, however, long-term side effects are less concerning and therefore should be balanced against potential benefit.

Supplemental Oxygen

Supplemental oxygen is often reflexively administered to patients presenting to the ED with dyspnea, particularly those patients suffering from hypoxia. While oxygen has been demonstrated to improve the QOL and longevity of patients with severe COPD, its use in patients with dyspnea at the EOL is more controversial [53, 54]. In one large randomized trial of patients with advanced illnesses and dyspnea, about 15% of whom had cancer, home oxygen resulted in no symptomatic improvements when compared to room air [55]. Surprisingly, in a prospective cohort study of patients admitted to a palliative care unit with advanced cancers, hypoxia and dyspnea showed little correlation. Furthermore, patients in this study had more dyspnea relief with opioid rather than oxygen administration [42]. As hypoxia is likely just one of the many factors contributing to dyspnea at the EOL, these results are somewhat predictable. Regardless, certain oxygen delivery devices like nasal cannula are minimally burdensome, so a time-limited trial in hypoxic, dyspneic patients can be considered, even those endorsing comfort-oriented goals. More caution should be used, however, when applying more burdensome devices like a non-rebreather mask, if the goals are comfort-oriented, as this may actually contribute to discomfort without clear evidence of efficacy.

Noninvasive Ventilation (NIV)/Mechanical Ventilation

NIV has been well studied and its efficacy well demonstrated in patients with CHF and COPD [56]. In recent years, this tool has been applied to a broader range of patients, even those with advanced cancers. A randomized trial evaluated the use of NIV when compared to oxygen (via venture or non-rebreather mask) for patients with advanced cancers presenting with acute respiratory failure. Patients receiving NIV had greater improvements in dyspnea scores and needed less opioids; however, about 10% of patients randomized to NIV discontinued therapy secondary to issues like mask intolerance (compared to none in the oxygen group). Mean survival was the same in both groups, 4-5 days; however, seemed to favor the use of NIV in those patients with concomitant hypercapnia [57]. In an observational study of patients with advanced cancer and respiratory failure, over half of patients treated with NIV actually survived their acute illness and were discharged alive from the hospital [58]. In a cohort of lung cancer patients receiving NIV, predictors of mortality included a new lung cancer diagnosis or progressive disease, multiorgan failure, and need of NIV as the firstline therapy for respiratory failure. The 1-month mortality for this small cohort was approximately 40% [59]. These studies suggest that for select patients with advanced cancer and a potentially reversible cause of acute dyspnea/respiratory failure, NIV can be a useful tool that improves symptoms and meaningfully prolongs life. In other patients, however, particularly those with respiratory failure secondary to progression of underlying disease, NIV may artificially prolong dying and worsen suffering at the EOL [60]. Prior to initiation of this therapy, it is recommended that clear, time-specific goals be established with the patient and/ or key decision-makers. For example, if there is no marked improvement in the patient's mental status and/or the mask seems to be causing discomfort in the next 24 h, NIV should be transitioned off, and opioids used exclusively to alleviate dyspnea.

Intubation is often considered, particularly in the ED, in advanced cancer patients with dyspnea and respiratory failure. Counseling patients and their families about the risks and benefits of this invasive intervention is critical. A recent study of cancer patients admitted to the ICU requiring ventilatory support demonstrated the importance of contextualizing the respiratory failure within a broader understanding of the overall illness. In the subgroup of patients with relapsed cancer and poor performance status (poor baseline function), hospital mortality was high, approaching 90% [61]. Trends over the last 10 years suggest more patients with metastatic cancer receiving mechanical ventilation, despite no changes in outcomes over this time frame [62]. Patients and families should be counseled that the use of mechanical ventilation is very unlikely to meaningfully prolong life in patients dying from an advanced cancer and is likely to lead to a burdensome death within the ICU setting.

High-Flow Nasal Cannula

High-flow nasal cannula (HFNC) has emerged a useful tool in managing hypoxemic respiratory failure in select patients. Heated and humidified oxygen is delivered at higher rates than would be possible via a traditional nasal cannula system. Unlike other advanced modalities like NIV and a nonrebreather mask, HFNC theoretically leads to improved comfort as patients can still communicate and eat and drink while wearing the device. In patients with respiratory failure and electing DNR/DNI status, HFNC seems to improve oxygenation and decrease the respiratory rate. When compared to other oxygen delivery modalities, however, there is no difference in dyspnea scores or morphine usage [63]. Similar to studies of NIV in patients with advanced cancer, mortality rates in these studies are high, ranging from 40 to 87%. In awake patients still able to engage with loved ones and/or interested in eating and drinking, this tool can be considered.

Management of Specific Conditions

Pleural Effusion

Malignant pleural effusion is a challenging entity in the emergency setting, particularly in symptomatic patients with guarded prognoses. It is a common condition, with an estimated 150,000 cases in the USA annually and a prevalence in advanced cancer of approximately 15% [64, 65]. Most pleural effusions will not become symptomatic until they reach over 500 cc in volume, and they can expand to over 2000 cc in volume. Practice guidelines for malignant effusions have traditionally recommended drainage with bedside thoracentesis, with consideration of pleurodesis-typically with talc-for prevention of recurrence and re-accumulation [66]. Talc pleurodesis, however, requires inpatient hospitalization and surgery, may be distressing and painful, and may lead to further complications including pneumothorax or empyema. Increasingly, indwelling pleural catheter drainage is being employed for either permanent or temporary management of re-accumulation, thus diverting patients away from the traditional and burdensome pleurodesis [64]. Compared to pleurodesis, patients with malignant effusions receiving an indwelling catheter spend fewer days in the hospital near the end of life [67]. Of note, palliative chemotherapy may actually benefit patients with recurrent effusions who have chemotherapeutic-responsive tumors [65]. This does not represent an emergency intervention but should be known when discussing the full range of therapeutic options with a family.

Functional status and patient prognosis should guide treatment, along with patient and family values and goals of care. In general, therapeutic thoracentesis alone is typically recommended in patients with a short expected prognosis or poor functional status, which allows for temporary evacuation of pleural fluid without necessitating hospital admission. Effusions may re-accumulate rapidly within days or slowly, on the order of months. There are no good predictors to help determine which patients are at a higher risk for rapid re-accumulation [64]. The potential for rapid recurrence is an important detail to share with patients' families who may have very different expectations of this procedure. Most, but not all, patients will experience relief in dyspnea following a thoracentesis, but given that dyspnea is multifactorial in advanced cancer, families should also understand that thoracentesis is not necessarily a definitive treatment for an individual patient's symptoms. For those patients with frequent recurrent pleural effusions or for those with predicted longer life expectancies, it may be more appropriate to refer for pleurodesis or tunneled catheter placement.

If performed in the emergency setting, thoracentesis should be done by an experienced provider utilizing ultrasound guidance. A retrospective study on 445 patients undergoing thoracentesis for malignant pleural effusion revealed a 0.97% pneumothorax rate with ultrasound guidance and 8.89% without [68]. Re-expansion pulmonary edema is a known potential complication and may occur if greater than 1.5 L are removed at once, although the incidence of this is uncommon, around 0.5% in a series of 185 cases [69, 70].

Anemia

Advanced cancer patients have a high prevalence of anemia, with studies suggesting up to 70% of these patients live with a hemoglobin concentration below 12 g/dL. While the majority of anemia in this population is of unclear etiology, many are thought to have anemia of chronic disease or nutritional deficiencies, notably of folic acid [71]. In a study aimed at establishing the cause of dyspnea in advanced cancer patients, up to 20% of enrollees were found to have a hemoglobin level which was low enough to have affected tissue perfusion [17]. In these situations, transfusion should be considered. Major elements impacting this decision include prognosis, functional status, and goals of care. There are risks and burden associated with transfusion, including additional time spent in the hospital, necessarily placement of intravenous access, and the risks of transfusion reaction, infection, or fluid overload. A prospective study of outpatient palliative care patients with cancer found that by using a transfusion cutoff of 8 g/dL, patients had a significant improvement in both self-reported dyspnea and fatigue, but these effects began to decrease approximately 2 weeks after transfusion [72]. The decision to transfuse should be left up to the treatment team in conjunction with the patient and family, though they should be aware that the subjective benefit in dyspnea relief is likely temporary and does carry some limited risk.

Oral Secretions

As death approaches, secretions pool in the posterior oropharynx, and patients become too weak to swallow or clear them. Anticholinergic agents are commonly used to address the noisy breathing that many patients experience as a result. Most patients are unconscious at this point and thus unlikely to be bothered by what is known as the "death rattle" [20]. Family members, however, can find the noise distressing, so treatment is often considered. Unfortunately, there is a lack of high-quality evidence on this topic to guide management. In a 2014 systematic review of interventions for noisy breathing near the end of life, only one small study was placebocontrolled, and the authors could find with no clear benefit found to any treatments evaluated [73].

For patients with intravenous access, glycopyrrolate 0.2 mg IV or atropine 0.1 mg IV can be administered. For those without IV access, drops of atropine 1% ophthalmic solution can be given by mouth [74]. More important than medication choice, however, is educating family and caregivers that this is a normal part of the dying process unlikely to be distressing to the patient. It is also an established prognostic sign that life expectancy is likely hours to days.

Tumor Burden

Lymphangitic carcinomatosis (LC), a condition involving hilar/mediastinal lymphatic inflammation secondary to malignant spread, affects between 6% and 8% of patients with intrathoracic metastases with the most common underlying primary tumors being breast, stomach, and lung [75]. LC manifests as nonspecific, nonproductive cough with associated dyspnea and may be definitively diagnosed by CT scan or bronchoscopy/biopsy [76]. As a late finding in advanced cancer, it carries a poor prognosis, with 50% survival at 3 months after first respiratory symptom. Corticosteroids have a palliative role by decreasing inflammation and should be considered in patients who carry this diagnosis [48]. In certain cases, palliative chemotherapy may also be offered and may give temporary improvement in symptoms over time.

Palliative Extubation

Establishing goals of care can be a time-intensive and arduous process. Oftentimes, a patient presents profoundly dyspneic to the ED alone or with EMS and with no ability to communicate their wishes about the use of life-sustaining treatments (LST). In these situations, unless emergent airway management appears futile, patients will be intubated and placed on mechanical ventilation. After medical stabilization, when it is possible to clarify a patient's wishes with family or supporting documentation, it may become clear that the patient's goals are inconsistent with mechanical ventilation. Appropriate management in these situations will vary—some families prefer to wait until they leave the ED into a more controlled environment (a hospital room or on a palliative care floor) prior to withdrawal of mechanical ventilation. In other situations, the emergency provider may withdraw support in the ED.

There may be hesitation on the part of the healthcare provider regarding the ramifications of withdrawal of LST in the ED. A public health survey of medical attending physicians in 1993 revealed that only 43% agreed that "there is no ethical difference between forgoing a life support measure and stopping it once it has been started" [77]. But from both an ethical and legal standpoint, there is no difference between these two actions [78]. The ethical principle of *autonomy*, which dictates that a patient has the right to make his or her own decisions, must be honored. Providers should seek documentation of the patient's wishes in the form of advance directives like a living will or POLST. When written directives are not present, surrogate decision-makers can use the principle of substituted judgment to make decisions that they feel would be in the best interest of the patient [79].

Prior to withdrawal of mechanical ventilation, family should be informed about the prognosis and potential outcomes following withdrawal. There is a common expectation that death is imminent after endotracheal tube removal, so appropriate counseling on the range of outcomes should be given. In a small cohort of patients terminally extubated in the ICU, mean time to death was 2–3 hours [80]. In another study of mechanically ventilated ICU patients who were terminally extubated, half died within 1 h of withdrawal, with the majority dying within 10 h [81]. Factors that were predictive of an earlier time to death included a high oxygen requirement (FIO₂ > 70%) and the use of vasopressors. Note that over half of this study group had been mechanically ventilated for over 10 days, and therefore these results may not be entirely reflective of the acute/emergency population. While most patients will live minutes to hours, families should be counseled that some can live for days.

Families should be allowed to make any necessary spiritual arrangement. Monitors and unnecessary equipment including blood pressure cuffs and pulse oximeters should be turned off and/or removed. It is best to keep IV access in place for rapid administration of sedatives if necessary.

There is debate about whether patients being extubated should be weaned from mechanical ventilation gradually, allowing the careful matching of opioids to symptoms, or extubated abruptly. In an uncontrolled study, when compared to immediate extubation, weaning seemed to lead to improved patient comfort [82]. If weaning is selected, the amount of pressure support and oxygen should be gradually decreased over 30 minutes prior to extubation. At the time of extubation, the tube cuff is deflated, and then the tube is removed, followed by cleaning and suctioning of oral secretions.

A dose of glycopyrrolate 0.2 mg IV can be given to minimize respiratory secretions, prior to extubation. As dyspnea is the primary symptom post-extubation, opioids are the first-line agents used during this procedure. Prior to extubation, an opioid should be given. Typical doses of morphine used are 4–6 mg. The goal of palliative medications is to minimize tachypnea and prevent agitation [79]. Families may be concerned with sedative administration, but similar to the management of undifferentiated dyspnea without airway management, provision of opioids after palliative extubation does not appear to have any effect on hastening death [83, 84]. Be available to the family, as they will often have questions during the dying process and may ask for frequent reexaminations of their loved one.

After the patient dies, ensure that the family is able to spend time around the bedside and provide bereavement support as needed.

Summary

Dyspnea in advanced cancer is distressing, complex, and often multifactorial. As patients approach the end of life, dyspnea becomes a common endpoint for multiple disease processes, with many factors simultaneously contributing to a patient's underlying shortness of breath. Many of these patients will seek care in the ED, necessitating emergency providers to have a strong understanding of the pathophysiology and management of terminal dyspnea. Above all, clear and open communication will help guide the diagnostic workup and treatment course for a patient with dyspnea near the end of life. Patient's goals may range from purely symptomatic treatment to aggressive extension of life, and therefore each patient will need to receive truly customized care from the onset of their ED visit.

Fortunately, while the neurophysiology of dyspnea is complex, the treatments are not. Opioids are the clear mainstay of symptomatic management. The proper titration of opioids will often be sufficient to adequately treat dyspnea at the end of life, but in certain situations a patient may require or request more intensive treatment. These interventions range from blood transfusions to noninvasive ventilation to bedside thoracentesis. By developing a keen understanding of the risks, benefits, and long-term outcomes of these interventions, the emergency provider is better able to equip the patient and family with the information they require to decide whether the intervention aligns with their goals and values. Acknowledgments The author wishes to acknowledge the contribution made by Dr. Trevor Pour, author of this chapter in the first edition upon which this revision is based.

References

- Parshall MB, Schwartzstein RM, Adams L, Banzett RB, Manning HL, Bourbeau J, et al. An official American Thoracic Society statement: update on the mechanisms, assessment, and management of dyspnea. Am J Respir Crit Care Med. 2012;185(4):435–52.
- Desbiens NA, Mueller-Rizner N, Connors AF, Wenger NS. The relationship of nausea and dyspnea to pain in seriously ill patients. Pain. 1997;71(2):149–56.
- Reuben DB, Mor V. Dyspnea in terminally ill cancer patients. Chest. 1986;89(2):234–6.
- Currow DC, Smith J, Davidson PM, Newton PJ, Agar MR, Abernethy AP. Do the trajectories of dyspnea differ in prevalence and intensity by diagnosis at the end of life? A consecutive cohort study. J Pain Symptom Manag. 2010;39(4):680–90.
- Ho SF, O'Mahony MS, Steward JA, Breay P, Buchalter M, Burr ML. Dyspnoea and quality of life in older people at home. Age Ageing. 2001;30(2):155–9.
- Edmonds P, Karlsen S, Khan S, Addington-Hall J. A comparison of the palliative care needs of patients dying from chronic respiratory diseases and lung cancer. Palliat Med. 2001;15(4):287–95.
- Booth S, Silvester S, Todd C. Breathlessness in cancer and chronic obstructive pulmonary disease: using a qualitative approach to describe the experience of patients and carers. Palliat Support Care. 2003;1(4):337–44.
- Davenport PW, Vovk A. Cortical and subcortical central neural pathways in respiratory sensations. Respir Physiol Neurobiol. 2009;167(1):72–86.
- Manning HL, Schwartzstein RM. Pathophysiology of dyspnea. N Engl J Med. 1995;333(23):1547–53.
- Lansing RW, Gracely RH, Banzett RB. The multiple dimensions of dyspnea: review and hypotheses. Respir Physiol Neurobiol. 2009;167(1):53–60.
- Peiffer C. Morphine-induced relief of dyspnea: what are the mechanisms? Am J Respir Crit Care Med. 2011;184(8):867–9.
- Moosavi SH, Golestanian E, Binks AP, Lansing RW, Brown R, Banzett RB. Hypoxic and hypercapnic drives to breathe generate equivalent levels of air hunger in humans. J Appl Physiol (1985). 2003;94(1):141–54.
- Clemens KE, Klaschik E. Dyspnoea associated with anxiety symptomatic therapy with opioids in combination with lorazepam and its effect on ventilation in palliative care patients. Support Care Cancer. 2011;19(12):2027–33.
- 14. Verhoef MJ, de Nijs E, Horeweg N, Fogteloo J, Heringhaus C, Jochems A, et al. Palliative care needs of advanced cancer patients in the emergency department at the end of life: an observational cohort study. Support Care Cancer. 2020;28(3):1097–107.
- Dudgeon DJ, Kristjanson L, Sloan JA, Lertzman M, Clement K. Dyspnea in cancer patients: prevalence and associated factors. J Pain Symptom Manag. 2001;21(2):95–102.
- Shreves A, Pour T. Emergency management of dyspnea in dying patients. Emerg Med Pract. 2013;15(5):1–19; quiz 20.
- Dudgeon DJ, Lertzman M. Dyspnea in the advanced cancer patient. J Pain Symptom Manag. 1998;16(4):212–9.
- Iserson KV. Withholding and withdrawing medical treatment: an emergency medicine perspective. Ann Emerg Med. 1996;28(1):51–4.

- Campbell ML, Templin T, Walch J. Patients who are near death are frequently unable to self-report dyspnea. J Palliat Med. 2009;12(10):881–4.
- Morita T, Ichiki T, Tsunoda J, Inoue S, Chihara S. A prospective study on the dying process in terminally ill cancer patients. Am J Hosp Palliat Care. 1998;15(4):217–22.
- Escalante CP, Martin CG, Elting LS, Price KJ, Manzullo EF, Weiser MA, et al. Identifying risk factors for imminent death in cancer patients with acute dyspnea. J Pain Symptom Manag. 2000;20(5):318–25.
- Zhang Z, Xu X. Lactate clearance is a useful biomarker for the prediction of all-cause mortality in critically ill patients: a systematic review and meta-analysis. Crit Care Med. 2014;42(9):2118–25.
- Zanobetti M, Poggioni C, Pini R. Can chest ultrasonography replace standard chest radiography for evaluation of acute dyspnea in the ED? Chest. 2011;139(5):1140–7.
- 24. Cibinel GA, Casoli G, Elia F, Padoan M, Pivetta E, Lupia E, et al. Diagnostic accuracy and reproducibility of pleural and lung ultrasound in discriminating cardiogenic causes of acute dyspnea in the emergency department. Intern Emerg Med. 2012;7(1):65–70.
- Henzler T, Barraza JM Jr, Nance JW Jr, Costello P, Krissak R, Fink C, et al. CT imaging of acute pulmonary embolism. J Cardiovasc Comput Tomogr. 2011;5(1):3–11.
- Tran QN. Role of palliative low-molecular-weight heparin for treating venous thromboembolism in patients with advanced cancer. Am J Hosp Palliat Care. 2010;27(6):416–9.
- Barnes H, McDonald J, Smallwood N, Manser R. Opioids for the palliation of refractory breathlessness in adults with advanced disease and terminal illness. Cochrane Database Syst Rev. 2016;3:CD011008.
- Bourke DL, Malit LA, Smith TC. Respiratory interactions of ketamine and morphine. Anesthesiology. 1987;66(2):153–6.
- Krajnik M, Jassem E, Sobanski P. Opioid receptor bronchial tree: current science. Curr Opin Support Palliat Care. 2014;8(3):191–9.
- Zebraski SE, Kochenash SM, Raffa RB. Lung opioid receptors: pharmacology and possible target for nebulized morphine in dyspnea. Life Sci. 2000;66(23):2221–31.
- Banzett RB, Adams L, O'Donnell CR, Gilman SA, Lansing RW, Schwartzstein RM. Using laboratory models to test treatment: morphine reduces dyspnea and hypercapnic ventilatory response. Am J Respir Crit Care Med. 2011;184(8):920–7.
- Vargas-Bermúdez A, Cardenal F, Porta-Sales J. Opioids for the Management of Dyspnea in cancer patients: evidence of the last 15 years--a systematic review. J Pain Palliat Care Pharmacother. 2015;29(4):341–52.
- 33. Woodcock AA, Gross ER, Gellert A, Shah S, Johnson M, Geddes DM. Effects of dihydrocodeine, alcohol, and caffeine on breathlessness and exercise tolerance in patients with chronic obstructive lung disease and normal blood gases. N Engl J Med. 1981;305(27):1611–6.
- Clemens KE, Klaschik E. Effect of hydromorphone on ventilation in palliative care patients with dyspnea. Support Care Cancer. 2008;16(1):93–9.
- Cohen MH, Anderson AJ, Krasnow SH, Spagnolo SV, Citron ML, Payne M, et al. Continuous intravenous infusion of morphine for severe dyspnea. South Med J. 1991;84(2):229–34.
- 36. Gamborg H, Riis J, Christrup L, Krantz T. Effect of intraoral and subcutaneous morphine on dyspnea at rest in terminal patients with primary lung cancer or lung metastases. J Opioid Manag. 2013;9(4):269–74.
- Viola R, Kiteley C, Lloyd NS, Mackay JA, Wilson J, Wong RK. The management of dyspnea in cancer patients: a systematic review. Support Care Cancer. 2008;16(4):329–37.

- King S, Forbes K, Hanks GW, Ferro CJ, Chambers EJ. A systematic review of the use of opioid medication for those with moderate to severe cancer pain and renal impairment: a European Palliative Care Research Collaborative opioid guidelines project. Palliat Med. 2011;25(5):525–52.
- Simon ST, Koskeroglu P, Gaertner J, Voltz R. Fentanyl for the relief of refractory breathlessness: a systematic review. J Pain Symptom Manag. 2013;46(6):874–86.
- Campbell ML. Psychometric testing of a respiratory distress observation scale. J Palliat Med. 2008;11(1):44–50.
- Ketwaroo GA, Cheng V, Lembo A. Opioid-induced bowel dysfunction. Curr Gastroenterol Rep. 2013;15(9):344.
- 42. Clemens KE, Quednau I, Klaschik E. Use of oxygen and opioids in the palliation of dyspnoea in hypoxic and non-hypoxic palliative care patients: a prospective study. Support Care Cancer. 2009;17(4):367–77.
- 43. Clemens KE, Quednau I, Klaschik E. Is there a higher risk of respiratory depression in opioid-naive palliative care patients during symptomatic therapy of dyspnea with strong opioids? J Palliat Med. 2008;11(2):204–16.
- 44. Clemens KE, Klaschik E. Symptomatic therapy of dyspnea with strong opioids and its effect on ventilation in palliative care patients. J Pain Symptom Manag. 2007;33(4):473–81.
- 45. Portenoy RK, Sibirceva U, Smout R, Horn S, Connor S, Blum RH, et al. Opioid use and survival at the end of life: a survey of a hospice population. J Pain Symptom Manag. 2006;32(6):532–40.
- 46. Simon ST, Higginson IJ, Booth S, Harding R, Weingärtner V, Bausewein C. Benzodiazepines for the relief of breathlessness in advanced malignant and non-malignant diseases in adults. Cochrane Database Syst Rev. 2016;10:CD007354.
- 47. Allcroft P, Margitanovic V, Greene A, Agar MR, Clark K, Abernethy AP, et al. The role of benzodiazepines in breathlessness: a single site, open label pilot of sustained release morphine together with clonazepam. J Palliat Med. 2013;16(7):741–4.
- Lin RJ, Adelman RD, Mehta SS. Dyspnea in palliative care: expanding the role of corticosteroids. J Palliat Med. 2012;15(7):834–7.
- Graves PR, Siddiqui F, Anscher MS, Movsas B. Radiation pulmonary toxicity: from mechanisms to management. Semin Radiat Oncol. 2010;20(3):201–7.
- Elsayem A, Bruera E. High-dose corticosteroids for the management of dyspnea in patients with tumor obstruction of the upper airway. Support Care Cancer. 2007;15(12):1437–9.
- Haywood A, Duc J, Good P, Khan S, Rickett K, Vayne-Bossert P, Hardy JR. Systemic corticosteroids for the management of cancerrelated breathlessness (dyspnoea) in adults. Cochrane Database Syst Rev. 2019;2(2):CD012704.
- Shih A, Jackson KC 2nd. Role of corticosteroids in palliative care. J Pain Palliat Care Pharmacother. 2007;21(4):69–76.
- Nocturnal Oxygen Therapy Trial Group. Continuous or nocturnal oxygen therapy in hypoxemic chronic obstructive lung disease: a clinical trial. Ann Intern Med. 1980;93(3):391–8.
- Eaton T, Lewis C, Young P, Kennedy Y, Garrett JE, Kolbe J. Longterm oxygen therapy improves health-related quality of life. Respir Med. 2004;98(4):285–93.
- 55. Abernethy AP, McDonald CF, Frith PA, Clark K, Herndon JE 2nd, Marcello J, et al. Effect of palliative oxygen versus room air in relief of breathlessness in patients with refractory dyspnoea: a doubleblind, randomised controlled trial. Lancet. 2010;376(9743):784–93.
- Weng CL, Zhao YT, Liu QH, Fu CJ, Sun F, Ma YL, et al. Metaanalysis: noninvasive ventilation in acute cardiogenic pulmonary edema. Ann Intern Med. 2010;152(9):590–600.
- 57. Nava S, Ferrer M, Esquinas A, Scala R, Groff P, Cosentini R, et al. Palliative use of non-invasive ventilation in end-of-life patients

with solid tumours: a randomised feasibility trial. Lancet Oncol. 2013;14(3):219–27.

- Cuomo A, Delmastro M, Ceriana P, Nava S, Conti G, Antonelli M, et al. Noninvasive mechanical ventilation as a palliative treatment of acute respiratory failure in patients with end-stage solid cancer. Palliat Med. 2004;18(7):602–10.
- Chen WC, Su VY, Yu WK, Chen YW, Yang KY. Prognostic factors of noninvasive mechanical ventilation in lung cancer patients with acute respiratory failure. PLoS One. 2018;13(1):e0191204.
- Quill CM, Quill TE. Palliative use of noninvasive ventilation: navigating murky waters. J Palliat Med. 2014;17(6):657–61.
- Azevedo LC, Caruso P, Silva UV, Torelly AP, Silva E, Rezende E, et al. Outcomes for patients with cancer admitted to the ICU requiring ventilatory support: results from a prospective multicenter study. Chest. 2014;146(2):257–66.
- 62. Gupta A, Das A, Tariq R, Bhulani N, Premnath N, Solanky D, et al. Trends in outcomes of patients with metastatic cancer undergoing intubation and mechanical ventilation: results of the national hospital discharge survey. J Natl Compr Cancer Netw. 2018;16(3):286–92.
- 63. Wilson ME, Mittal A, Dobler CC, Curtis JR, Majzoub AM, Soleimani J, et al. High-flow nasal cannula oxygen in patients with acute respiratory failure and do-not-intubate or do-not-resuscitate orders: a systematic review. J Hosp Med. 2020;15(2):101–6.
- 64. Thomas R, Francis R, Davies HE, Lee YC. Interventional therapies for malignant pleural effusions: the present and the future. Respirology. 2014;19(6):809–22.
- Antony VB, Loddenkemper R, Astoul P, Boutin C, Goldstraw P, Hott J, et al. Management of malignant pleural effusions. Eur Respir J. 2001;18(2):402–19.
- 66. Dy SM, Lorenz KA, Naeim A, Sanati H, Walling A, Asch SM. Evidence-based recommendations for cancer fatigue, anorexia, depression, and dyspnea. J Clin Oncol. 2008;26(23):3886–95.
- 67. Thomas R, Fysh ETH, Smith NA, Lee P, Kwan BCH, Yap E, et al. Effect of an indwelling pleural catheter vs talc pleurodesis on hospitalization days in patients with malignant pleural effusion: the AMPLE randomized clinical trial. JAMA. 2017;318(19):1903–12.
- 68. Cavanna L, Mordenti P, Berte R, Palladino MA, Biasini C, Anselmi E, et al. Ultrasound guidance reduces pneumothorax rate and improves safety of thoracentesis in malignant pleural effusion: report on 445 consecutive patients with advanced cancer. World J Surg Oncol. 2014;12:139.
- Feller-Kopman D, Berkowitz D, Boiselle P, Ernst A. Large-volume thoracentesis and the risk of reexpansion pulmonary edema. Ann Thorac Surg. 2007;84(5):1656–61.
- 70. Roberts ME, Neville E, Berrisford RG, Antunes G, Ali NJ. Management of a Malignant Pleural Effusion: British Thoracic

Society pleural disease guideline 2010. Thorax. 2010;65(Suppl 2):32-40.

- Dunn A, Carter J, Carter H. Anemia at the end of life: prevalence, significance, and causes in patients receiving palliative care. J Pain Symptom Manag. 2003;26(6):1132–9.
- Mercadante S, Ferrera P, Villari P, David F, Giarratano A, Riina S. Effects of red blood cell transfusion on anemia-related symptoms in patients with cancer. J Palliat Med. 2009;12(1):60–3.
- Lokker ME, van Zuylen L, van der Rijt CC, van der Heide A. Prevalence, impact, and treatment of death rattle: a systematic review. J Pain Symptom Manag. 2014;47(1):105–22.
- Bickel K, Arnold RM. Death rattle and oral secretions—second edition #109. J Palliat Med. 2008;11(7):1040–1.
- 75. Bruce DM, Heys SD, Eremin O. Lymphangitis carcinomatosa: a literature review. J R Coll Surg Edinb. 1996;41(1):7–13.
- Storck K, Crispens M, Brader K. Squamous cell carcinoma of the cervix presenting as lymphangitic carcinomatosis: a case report and review of the literature. Gynecol Oncol. 2004;94(3):825–8.
- Solomon MZ, O'Donnell L, Jennings B, Guilfoy V, Wolf SM, Nolan K, et al. Decisions near the end of life: professional views on life-sustaining treatments. Am J Public Health. 1993;83(1):14–23.
- Melltorp G, Nilstun T. The difference between withholding and withdrawing life-sustaining treatment. Intensive Care Med. 1997;23(12):1264–7.
- Bookman K, Abbott J. Ethics seminars: withdrawal of treatment in the emergency department – when and how? Acad Emerg Med. 2006;13(12):1328–32.
- Mazer MA, Alligood CM, Wu Q. The infusion of opioids during terminal withdrawal of mechanical ventilation in the medical intensive care unit. J Pain Symptom Manag. 2011;42(1):44–51.
- Huynh TN, Walling AM, Le TX, Kleerup EC, Liu H, Wenger NS. Factors associated with palliative withdrawal of mechanical ventilation and time to death after withdrawal. J Palliat Med. 2013;16(11):1368–74.
- 82. Robert R, Le Gouge A, Kentish-Barnes N, Cottereau A, Giraudeau B, Adda M, et al. Terminal weaning or immediate extubation for withdrawing mechanical ventilation in critically ill patients (the ARREVE observational study). Intensive Care Med. 2017;43(12):1793–807.
- Chan JD, Treece PD, Engelberg RA, Crowley L, Rubenfeld GD, Steinberg KP, et al. Narcotic and benzodiazepine use after withdrawal of life support: association with time to death? Chest. 2004;126(1):286–93.
- Edwards MJ. Opioids and benzodiazepines appear paradoxically to delay inevitable death after ventilator withdrawal. J Palliat Care. 2005;21(4):299–302.

Cardiopulmonary Resuscitation

Mohammad Adrian Hasdianda and Kei Ouchi

Introduction

Cardiopulmonary resuscitation (CPR) can be a lifesaving intervention; however, the indiscriminate use of CPR among unselected populations, and particularly among those with cancer, confers beneficial outcomes (e.g., survival to hospital discharge) on only a small proportion of patients [1–12]. In the USA, CPR is routinely provided to those suffering cardiac arrest without their consent. Only when patients give caregivers explicit instructions to withhold CPR is it not performed [1, 13, 14].

In 1960, Kouwenhoven et al. [14, 15] first described closedchest massage, intended for administration to otherwise "healthy patients" with reversible conditions who experienced sudden and unexpected cardiorespiratory arrest. Today, despite the near universal application of CPR, in most cases and particularly among cancer patients, CPR merely prolongs the dying process [2, 12, 16-22]. During the last 15 years, researchers have determined that cancer patients have a particularly low rate of return of spontaneous circulation (ROSC) and survival to hospital discharge (SHD) after CPR [1-11]. SHD rates are substantially better for in-hospital CPR compared to out-of-hospital CPR among all comers. SHD rates for out-of-hospital CPR and in-hospital CPR in an unselected population undergoing CPR are 1-10% and 15%, respectively [2, 16-22], but for cancer populations it is <6% [9, 10]. Even among patients with stage IV cancers who survive in-hospital CPR, the median survival after discharge is only 22 days [23]. Cardiopulmonary arrest is the common final pathway for most terminal diseases, including cancer [9, 10]. Despite the low rates of SHD and short survival after discharge, CPR may be aligned to patient's goals of care. More information is needed to better understand the quality of life for survivors and under what conditions CPR is appropriate.

An increased emphasis on palliative and supportive care for cancer patients improves quality of life [24–26]. Palliative

M. A. Hasdianda · K. Ouchi (🖂)

Department of Emergency Medicine, Brigham and Women's Hospital, Boston, MA, USA e-mail: kouchi@partners.org services incorporated in planning and executing therapeutic interventions hold the promise that CPR might be used more selectively. A more selective approach to initiating CPR, incorporating the patient's goals of care, should lead to increases in goal-concordant care.

In this chapter, we discuss how to determine whether CPR is aligned with goals of care among patients with cancer, explore palliative and supportive care resources, and describe CPR outcomes in tertiary cancer centers. We will also examine the issue of family-witnessed resuscitation.

What Outcomes After CPR Are Acceptable to Patients?

Emergency physicians understand that when patients with cancer arrive in the emergency department (ED), many have an inaccurate understanding of their illness and prognosis. Studies show that 69% of patients with stage IV lung cancer and 81% of those with colorectal cancer did not report understanding that chemotherapy was not at all likely to cure their cancer [27]. When asked about CPR, only 2.7% of hospitalized older patients with end-stage cancer or other advanced diseases understood that actual CPR success rates were <10%. Only 11% of these patients could describe more than two components involved in CPR [28].

To determine whether CPR is indicated for ED cancer patients with limited understanding of both their prognosis and CPR, emergency physicians should focus on a patientcentered prognosis (i.e., what patients with serious, lifelimiting illnesses would consider good outcomes). In a national survey study of physicians and seriously ill patients, a number of end-of-life items were rated as significantly more important to patients than physicians, including being mentally aware, having made funeral arrangements, feeling that one's life was complete, not being a burden, being able to help others, coming to peace with God, and praying [29]. Furthermore, $\geq 60\%$ of seriously ill older adults consider inability to "get out of bed" or "rely on a breathing machine

© Springer Nature Switzerland AG 2021 K. H. Todd et al. (eds.), *Oncologic Emergency Medicine*, https://doi.org/10.1007/978-3-030-67123-5_58



to live" as equal to or "worse than death" [30]. The vast majority (87%) of seriously ill, hospitalized, older adults express that they would even trade 1 year of a 5-year lifespan to avoid dying in an ICU on life support [31]. Considering prognosis and likely functional outcomes, emergency physicians must determine what patients would consider an acceptable quality of life should they survive CPR.

How Should Emergency Physicians Make Recommendations Regarding CPR?

To make an empathic and goal-concordant recommendation around CPR, integrate the patient's baseline function and values with knowledge of the patient's prognosis. Ask yourself, "In the best-case scenario, would this patient be able to achieve the minimal quality of life worth living for after CPR?" If this answer is a clear "no" or likely outcome would be considered "worse than death" for the patient [30], emergency physicians can confidently make a recommendation against CPR. If the answer is unclear (e.g., the surrogate may not know the minimal quality of life that patient would consider acceptable to live) or likely outcome would be an acceptable quality of life worth living for, emergency physicians can make a recommendation for CPR (Fig. 58.1). The language used to make the recommendation should reflect how the recommendation was made (e.g., "based on what I heard about you, I would recommend CPR"). Emphasize what you will do (e.g., "focus on ensuring comfort"). Consider explaining why you would not recommend certain therapies in the context of the baseline function and values [32].

Palliative and Supportive Care Interventions

The goal of palliative and supportive care is to prevent or treat, as early as possible, the symptoms of the disease, side effects caused by treatment, and psychological, social, and

Fig. 58.1 CPR

recommendation based on quality of life acceptable to patients. (From Ouchi et al. [33], with permission Elsevier) spiritual problems related to the disease or its treatment. Palliative care focuses on the assessment and management of physical and psychosocial distress of patients with advanced cancer as well as family support and advance care planning [24, 25]. Patients make better-informed decisions with less distress when physical and emotional symptoms are controlled [26]. Furthermore, the manner in which physicians communicate about CPR is important. Of patients with advanced cancer randomly assigned to learn about CPR by either viewing a video decision support tool or listening to a verbal narrative, 79% of those viewing the video opted out of CPR, while only 51% receiving a verbal description decided against CPR [34].

Palliative care is not only pertinent to the end of life; rather, it can and should be provided early in the cancer journey. When engaged early, these services improve quantity and quality of life concurrently with the oncologic care model [26]. This model enables supportive/palliative care to be integrated into the collaborative model that exists among surgical, radiation, and medical oncologists as the fourth pillar of comprehensive cancer care. This multidisciplinary approach allows additional consultants the opportunity to participate in care including, but not limited to, pain specialists; psychiatrists for emotional distress, depression, and anxiety; pulmonologists for relief of bronchial obstruction; and psychosocial interventionists for end-of-life issues [9– 11, 24–26]. Incorporating end-of-life education into the dayto-day management of these patients is essential.

Comprehensive Cancer Centers

In one comprehensive cancer center, between the periods 2002–2007 and 2008–2012, rates of ROSC after CPR showed small and statistically insignificant improvements, while SHD outcomes did not change [35] (Fig. 58.2). Any trend toward improved ROSC outcomes that may exist could result from improvements in CPR quality [36, 37].



Fig. 58.2 Survival to discharge among 126 cancer patients undergoing cardiopulmonary resuscitation (CPR) in a comprehensive cancer center comparing group 1 (2002–2007) and group 2 (2008–2012). (From Miller et al. [35] with permission SpringerPlus [CC by 4.0])



The lack of improvement in SHD may suggest that CPR in cancer patients continues to be performed on an unselected cancer population, rather than being targeting toward subsets of cancer patients who are more likely to receive benefit.

Reisfield et al. [11] reported in a 2006 meta-analysis that among 1118 patients with cancer receiving CPR, the rate of SHD was 7.1% among those with solid tumor vs. 2% for those with hematologic cancer. Consistent with these data, Hwang et al. [10] reported that during 2000–2002, of 41 cancer patients receiving CPR in their ED, only 6 of 33 patients (18%) with solid tumors and 1 of 8 (13%) with hematologic malignancies were ultimately discharged alive to another facility.

Future research could identify the specific type(s) of cancer that could benefit from CPR. End-of-life and advanced care planning may contribute to improved outcomes of CPR by allowing those terminally ill to avoid ineffective terminal interventions [38]. Improved advance care planning communication and/or documentation and family education can be exploited to improve CPR outcomes in both cancer and noncancer populations. These interventions allow patients and loved ones to avoid transportation to the hospital and CPR [1, 39–41]. As cancer

populations increase in the USA, discussions about end-oflife issues will become particularly relevant with regard to economic impact [12].

Family-Witnessed Resuscitation

As noted, survival outcomes measured by improved quality of life are particularly dismal after CPR for cancer patients. Even though the statistics are unfavorable, the decision to resuscitate is often multifactorial. Ideally, end-of-life planning should occur early in the course of treatment so that patient wishes will be clear and based on an adequate consideration of alternate treatment plans. Indeed, planning should occur at all stages of disease.

An important topic particularly relevant to emergency care is that of family-witnessed resuscitation (FWR). According to Boyd, FWR is "the process of active 'medical' resuscitation in the presence of family members [42]. The concept of having family members present during resuscitative efforts was introduced in the 1980s when Foote Hospital in Michigan began promoting FWR [43]. In 1992, Hanson and Strawser presented initial research data on this topic [44]. Twenty years later, in 2000, the American Heart Association published guidelines recommending that family members be allowed to witness CPR. At the time, there was a lack of research to provide quantitative proof that FWR was beneficial. Another 20 years later, FWR remains the exception in EDs internationally [45].

The question of whether family members or loved ones desire to be present during resuscitation, or whether the emergency physician should consider offering this choice, remains an important policy issue for the ED. Planning and establishing procedures for FWR should occur well before the need to make such decisions arise [46]. We will discuss the pros and cons based on perspectives of all parties that may be present during resuscitation and ethical contexts regarding this issue. We suggest conditions under which FWR should be considered based on a review of existing research.

Perspectives of Family Members

Family members of both pediatric and adult patients play an increasingly larger role as caregivers given the shift in emphasis of care from the hospital to outpatient setting. Oftentimes, they initiate an emergency response in the pre-hospital setting if their loved one is in distress. Given that family members often participate in treatment planning and day-to-day care, it is not uncommon for them to wish to be present with their loved one during the final moments of life, including resuscitation attempts.

Multiple studies focusing on the attitudes of family members reveal that most prefer to be present during resuscitation when given the opportunity. One 1982 survey assessing bereaved family members in Michigan found that 72% preferred to be present during resuscitation attempts. Other studies confirm the strong desire of family members to be present and that those participating in FWR would recommend the same to other families [47–52].

Despite the strong preference of family members to participate in FWR, the practice remains uncommon. Hospital staff have traditionally excluded nonmedical personnel because witnessing the resuscitation was thought to cause emotional distress. Multiple studies counter this belief and indicate that family members do not suffer negative psychological consequences after witnessing resuscitation. In 2013, Jabre et al. [53] performed a 1-year post-resuscitation study of 408 family members measuring symptoms of posttraumatic stress disorder (PTSD), anxiety, depression, and complications of grief. Family members witnessing resuscitation appeared to suffer less PTSD-related symptoms [53, 54]. A common belief is that participating in FWR allows for a sense of closure after family members witness that everything possible had been done to increase the chance of survival [49, 52, 55, 56].

Perspectives of Healthcare Workers

With endorsement by the Resuscitation Council UK, the European Resuscitation Council, and the American Heart Association, FWR is more frequently performed in the ED. In a survey of 162 UK EDs with a mean patient volume of 47,000 patients per year, 79% (128 EDs) reported allowing adult patients to have FWR [57]. However, FWR is not universally supported by healthcare workers despite studies suggesting that it is beneficial to family members. A survey of 132 ED staff members at a hospital in Singapore found that 80% of doctors and 78% of nurses actually disapproved FWR. Of those surveyed, 32% of doctors and 24% of nurses had received requests from relatives to be present during resuscitation within the past 6 months [58]. Another survey of 100 healthcare professionals at an academic medical center in the USA showed a contrasting result, in that the majority of staff members (77%) were in favor of allowing FWR [59].

Attitudes of healthcare workers toward FWR differ by discipline, patient age, and practice environment. Nurses are more open to FWR than doctors, and family members tend to approach a nurse rather than physician asking permission to witness resuscitation of their loved one [60, 61]. Those caring for younger patients and their families are generally more positive toward FWR. One-third of pediatricians surveyed would allow family presence during CPR, and almost two-thirds with FWR experience would allow this to happen again. In a study comparing pediatric vs. adult pulmonologists, pediatric pulmonologists were far more accepting of FWR [62]. The nature of pediatric care may allow staff to be more accepting of FWR as a right for parents who are legal guardians for their children [63]. A small South African qualitative study revealed that ED staff disliked FWR as they believed it to be harmful for the witnesses as well as a threat to the staff [45].

Practice environment also influences the attitudes of healthcare workers. Staff at urban hospitals are less supportive of FWR than staff at suburban hospitals [64]. Macy suggests that logistics may play a role given that urban hospitals may have inadequate resuscitation space per patient volume. Urban hospitals have smaller staff/patient ratios which may cause staff to feel that family members' presence is a distraction. This study found that patient/personnel ethnicity had no significant effect on overall attitudes toward FWR by medical staff [64]. A cross-sectional study conducted at an American community hospital suggested that critical care nurses who had experience FWR had a trend toward more positive attitudes about the benefits and outcomes of FWR, though the findings were not statistically significant [65].

In general, healthcare professionals opposing FWR express a common concern that witnessing a loved one

undergoing aggressive resuscitation predisposes family members to additional psychological burden [63]. Staff report that witnessing resuscitation may cause family members to suffer flashbacks and other signs of posttraumatic psychological trauma. The reality of what may occur during resuscitation differs than what is usually depicted in movies or on television, and nonmedical personnel may not be mentally prepared to witness a real resuscitation.

A second concern expressed by many healthcare workers is that FWR negatively impacts staff performance. The presence of family members may cause additional stress to healthcare workers, especially those less experienced with code blue situations [66]. Resuscitation is a stressful event, and coping mechanisms, such as humor or detachment from a patient, may certainly be deemed inappropriate by family members. Unrealistic expectations may also exist as family members are not expected to understand resuscitation procedures and may interfere or interrupt resuscitation efforts. Resuscitation may also be inappropriately extended beyond usual time limits based on family presence. An increase in potential for litigation is also cited as an issue given that relatives in the room may be at risk for needle sticks or being injured by a piece of equipment [67, 68]. These findings contrast with the result of several other studies [54, 65].

Given these concerns, most healthcare workers recommend FWR in a controlled setting with protocols in place and specifically trained personnel to accompany family members. Hospital staff would also require additional training which would add to hospital expenses [69, 70]. After receiving education, it is interesting to note that in some studies, staff felt that witnessed resuscitation may indeed lessen the risk of lawsuits and complaints given that family members witnessed firsthand that everything was done for their loved one. Communication about the death of the relative was also found to be easier [58, 63, 71, 72].

Perspectives of Patients

Only 10–15% of patients receiving CPR following cardiac arrest in the hospital survive to be discharged. Therefore, fewer studies exist which examine FWR from a patient perspective. Albarran performed face-to-face interviews with 21 resuscitation survivors, and although the study was statistically insignificant, it suggests that patients who survive resuscitation favor having family members present [73]. Patients would like the opportunity to be asked to approve FWR and were not as concerned about confidentiality issues. Another pilot study by Robinson et al. consisting of three patients surviving resuscitation reported that their confidentiality and dignity were not compromised by FWR [71].

Ethical Considerations

Patients are rarely asked about their preference for familywitnessed resuscitation when preparing advanced directives. The lack of documented consent creates medicolegal implications for FWR because of the potential for breaching patient confidentiality. Acceptable guidelines for pediatric patients do not transfer to adults, given that family members have no legal rights to care for their adult relatives. For adults, permission must be granted from the patient before discussing medical care with relatives. Even if a patient is unconscious, these rights are still present; thus, an assumption cannot be made that all patients would give automatic consent for FWR [63]. Consequences must be considered if a patient survives the resuscitation and is not happy with the decision, though concerns regarding litigations from FWR have not been demonstrated [74].

Guidelines

When considering FWR, the interests of patients, family members, and staff must be taken into consideration, and this is difficult during an emergency situation. Optimally, FWR should be discussed with patients prior to potential resuscitations, e.g., during advance directive planning. If a hospital decides to incorporate FWR, it is recommended that an ED-specific protocol be established. Training of specific ancillary staff should occur so that a qualified staff member can accompany the relative at all times during the resuscitation. The role of this staff member will be to debrief the loved ones prior to entering the resuscitation room, to answer questions during the resuscitation, to escort the relative out of the room if necessary, and to help debrief the relative after resuscitation. Education is desired for the multidisciplinary resuscitation team on what to expect with FWR so that any fear or apprehension can be addressed. If a protocol for FWR is in place and staff are educated, they will less likely deny family member requests to be with their loved one during resuscitation, and family members can be allowed near or even touch their loved one during the process with less concern interfering with resuscitation procedures [69, 75].

Conclusions

In this chapter, we have reviewed the literature pertaining to CPR on patients with cancer. To improve the current status, more teaching of palliative and supportive care to maximize quality of life concurrently with disease-oriented therapies is needed. Despite well-intentioned resuscitation efforts over the last 40 years, the survival outcomes for those with cancer particularly metastatic cancer have not improved [11, 38,

41]. FWR should be considered in the ED; however, the ultimate impact of FWR on family members and loved ones remains uncertain.

References

- Varon J, Marik PE. Cardiopulmonary resuscitation in patients with cancer. Am J Hosp Palliat Care. 2007;24(3):224–9.
- Wenzel V, Krismer AC, Arntz HR, Sitter H, Stadlbauer KH, Lindner KH. A comparison of vasopressin and epinephrine for out-of-hospital cardiopulmonary resuscitation. N Engl J Med. 2004;350(2):105–13.
- Taran A, Guarino M, Kolm P, Petrelli N. Cardiopulmonary resuscitation inpatient outcomes in cancer patients in a large community hospital. Del Med J. 2012;84(4):117–21.
- Leak A, Mayer DK, Wyss A, Travers D, Waller A. Why do cancer patients die in the emergency department? An analysis of 283 deaths in NC EDs. Am J Hosp Palliat Care. 2013;30(2): 178–82.
- Fu S, Barber FD, Naing A, et al. Advance care planning in patients with cancer referred to a phase I clinical trials program: the MD Anderson Cancer Center experience. J Clin Oncol. 2012;30(23):2891–6.
- Tan TS, Jatoi A. End-of-life hospital costs in cancer patients: do advance directives or routes of hospital admission make a difference? Oncology. 2011;80(1–2):118–22.
- Ho TH, Barbera L, Saskin R, Lu H, Neville BA, Earle CC. Trends in the aggressiveness of end-of-life cancer care in the universal health care system of Ontario, Canada. J Clin Oncol. 2011;29(12):1587.
- Fu S, Hong DS, Naing A, et al. Outcome analyses after the first admission to an intensive care unit in patients with advanced cancer referred to a phase I clinical trials program. J Clin Oncol. 2011;29(26):3547–52.
- Myrianthefs P, Batistaki C, Baltopoulos G. Cardiopulmonary resuscitation in end-stage cancer patients. J BUON. 2010;15(1): 25–8.
- Hwang JP, Patlan J, De Achaval S, Escalante CP. Survival in cancer patients after out-of-hospital cardiac arrest. Support Care Cancer. 2010;18(1):51–5.
- Reisfield GM, Wallace SK, Munsell MF, Webb FJ, Alvarez ER, Wilson GR. Survival in cancer patients undergoing in-hospital cardiopulmonary resuscitation: a meta-analysis. Resuscitation. 2006;71(2):152–60.
- Wallace S, Ewer MS, Price KJ, Feeley TW. Outcome and cost implications of cardiopulmonary resuscitation in the medical intensive care unit of a comprehensive cancer center. Support Care Cancer. 2002;10(5):425–9.
- 13. Hansen-Flaschen JH. When life support is futile. Chest. 1991;100(5):1191–2.
- Kouwenhoven WB, Jude JR, Knickerbocker GG. Closed-chest cardiac massage. JAMA. 1960;173(10):1064–7.
- Bernard SA, Gray TW, Buist MD, et al. Treatment of comatose survivors of out-of-hospital cardiac arrest with induced hypothermia. N Engl J Med. 2002;346(8):557–63.
- Schneider AP, Nelson DJ, Brown DD. In-hospital cardiopulmonary resuscitation: a 30-year review. J Am Board Fam Pract. 1993;6(2):91–101.
- Bedell SE, Delbanco TL, Cook EF, Epstein FH. Survival after cardiopulmonary resuscitation in the hospital. N Engl J Med. 1983;309(10):569–76.
- Roberts D, Landolfo K, Light RB, Dobson K. Early predictors of mortality for hospitalized patients suffering cardiopulmonary arrest. Chest. 1990;97(2):413–9.

- Berger R, Kelley M. Survival after in-hospital cardiopulmonary arrest of noncritically ill patients: a prospective study. Chest. 1994;106(3):872–9.
- Stiell IG, Hebert PC, Weitzman BN, Wells GA, Raman S, Stark RM, et al. High-dose epinephrine in adult cardiac arrest. N Engl J Med. 1992;327(15):1045–50.
- Brown CG, Martin DR, Pepe PE, Stueven H, Cummins RO, Gonzalez E, Jastremski M. A comparison of standard-dose and high-dose epinephrine in cardiac arrest outside the hospital. The Multicenter High-Dose Epinephrine Study Group. N Engl J Med. 1992;327(15):1051–5.
- Becker LB, Ostrander MP, Barrett J, Kondos GT. Outcome of CPR in a large metropolitan area—where are the survivors? Ann Emerg Med. 1991;20(4):355–61.
- 23. Lee MR, Yu KL, Kuo HY, Liu TH, Ko JC, Tsai JS, Wang JY. Outcome of stage IV cancer patients receiving in-hospital cardiopulmonary resuscitation: a population-based cohort study. Sci Rep. 2019;9(1):9478.
- 24. Fadul N, Elsayem A, Palmer JL, Del Fabbro E, Swint K, Li Z, et al. Supportive versus palliative care: what's in a name?: a survey of medical oncologists and midlevel providers at a comprehensive cancer center. Cancer. 2009;115(9):2013–21.
- Hui D, Elsayem A, De la Cruz M, Berger A, Zhukovsky DS, Palla S, et al. Availability and integration of palliative care at US cancer centers. JAMA. 2010;303(11):1054–61.
- Bruera E, Hui D. Integrating supportive and palliative care in the trajectory of cancer: establishing goals and models of care. J Clin Oncol. 2010;28(25):4013–7.
- Wigginton JG, Miller AH, Benitez FL, Pepe PE. Mechanical devices for cardiopulmonary resuscitation. Curr Opin Crit Care. 2005;11(3):219–23.
- Heyland DK, Frank C, Groll D, Pichora D, Dodek P, Rocker G, Gafni A. Understanding cardiopulmonary resuscitation decision making: perspectives of seriously ill hospitalized patients and family members. Chest. 2006;130(2):419–28.
- Steinhauser KE, Christakis NA, Clipp EC, McNeilly M, McIntyre L, Tulsky JA. Factors considered important at the end of life by patients, family, physicians, and other care providers. JAMA. 2000;284(19):2476–82.
- Rubin EB, Buehler AE, Halpern SD. States worse than death among hospitalized patients with serious illnesses. JAMA Intern Med. 2016;176(10):1557–9.
- Rubin EB, Buehler A, Halpern SD. Seriously ill patients' willingness to trade survival time to avoid high treatment intensity at the end of life. JAMA Intern Med. 2020;180(6):907–9.
- Jacobsen J, Blinderman C, Alexander Cole C, Jackson V. "I'd recommend ..." how to incorporate your recommendation into shared decision making for patients with serious illness. J Pain Symptom Manag. 2018;55(4):1224–30.
- Ouchi K, Lawton AJ, Bowman J, Bernacki R, George N. Managing code status conversations for seriously ill older adults in respiratory failure. Ann Emerg Med. 2020;76(6):751–6. https://doi. org/10.1016/j.annemergmed.2020.05.039.
- 34. Volandes AE, Paasche-Orlow MK, Mitchell SL, El-Jawahri A, Davis AD, Barry MJ, et al. Randomized controlled trial of a video decision support tool for cardiopulmonary resuscitation decision making in advanced cancer. J Clin Oncol. 2013;31(3): 380–6.
- Miller AH, Sandoval M, Wattana M, Page VD, Todd KH. Cardiopulmonary resuscitation outcomes in a cancer center emergency department. Springerplus. 2015;4(1):106. https://doi. org/10.1186/s40064-015-0884-z.
- 36. Wolcke BB, Mauer DK, Schoefmann MF, Teichmann H, Provo TA, Lindner KH, et al. Comparison of standard cardiopulmonary resuscitation versus the combination of active compressiondecompression cardiopulmonary resuscitation and an inspiratory

impedance threshold device for out-of-hospital cardiac arrest. Circulation. 2003;108(18):2201-5.

- Hypothermia after Cardiac Arrest Study Group. Mild therapeutic hypothermia to improve the neurologic outcome after cardiac arrest. N Engl J Med. 2002;346(8):549–56.
- Sculier JP. Cardiopulmonary resuscitation in cancer patients: indications and limits. Clin Intensive Care. 1995;6(2):72–5.
- Gleeson K, Wise S. The do-not-resuscitate order: still too little too late. Arch Intern Med. 1990;150(5):1057–60.
- Eliott JA, Olver IN. The implications of dying cancer patients' talk on cardiopulmonary resuscitation and do-not-resuscitate orders. Qual Health Res. 2007;17(4):442–55.
- Hwang JP, Smith ML, Flamm AL. Challenges in outpatient end-of-life care: wishes to avoid resuscitation. J Clin Oncol. 2004;22(22):4643–5.
- 42. Boyd R. Witnessed resuscitation by relatives. Resuscitation. 2000;43(3):171-6.
- Doyle CJ, Post H, Burney RE, Maino J, Keefe M, Rhee KJ. Family participation during resuscitation: an option. Ann Emerg Med. 1987;16(6):673–5.
- Hanson C, Strawser D. Family presence during cardiopulmonary resuscitation: Foote Hospital emergency department's nine-year perspective. J Emerg Nurs. 1992;18(2):104–6.
- 45. Goodenough T, Brysiewicz P. Witnessed resuscitation-exploring the attitudes and practices of the emergency staff working in Level I Emergency Departments in the province of KwaZulu-Natal. Curationis. 2003;26(2):56–63.
- 46. Sak-Dankosky N, Andruszkiewicz P, Sherwood PR, Kvist T. Health care professionals' concerns regarding in-hospital family-witnessed cardiopulmonary resuscitation implementation into clinical practice. Nurs Crit Care. 2018;23(3):134–40.
- 47. Dingeman RS, Mitchell EA, Meyer EC, Curley MA. Parent presence during complex invasive procedures and cardiopulmonary resuscitation: a systematic review of the literature. Pediatrics. 2007;120(4):842–54.
- 48. Dudley NC, Hansen KW, Furnival RA, Donaldson AE, Van Wagenen KL, Scaife ER. The effect of family presence on the efficiency of pediatric trauma resuscitations. Ann Emerg Med. 2009;53(6):777–84, e773.
- McGahey-Oakland PR, Lieder HS, Young A, Jefferson LS. Family experiences during resuscitation at a children's hospital emergency department. J Pediatr Health Care. 2007;21(4):217–25.
- 50. Mortelmans LJ, Van Broeckhoven V, Van Boxstael S, De Cauwer HG, Verfaillie L, Van Hellemond PL, et al. Patients' and relatives' view on witnessed resuscitation in the emergency department: a prospective study. Eur J Emerg Med. 2010;17(4):203–7.
- Piira T, Sugiura T, Champion G, Donnelly N, Cole A. The role of parental presence in the context of children's medical procedures: a systematic review. Child Care Health Dev. 2005;31(2): 233–43.
- Tinsley C, Hill JB, Shah J, Zimmerman G, Wilson M, Freier K, Abd-Allah S. Experience of families during cardiopulmonary resuscitation in a pediatric intensive care unit. Pediatrics. 2008;122(4):e799–804.
- 53. Jabre P, Tazarourte K, Azoulay E, Borron SW, Belpomme V, Jacob L, et al. Offering the opportunity for family to be present during cardiopulmonary resuscitation: 1-year assessment. Intensive Care Med. 2014;40(7):981–7.
- 54. Jabre P, Belpomme V, Azoulay E, Jacob L, Bertrand L, Lapostolle F, et al. Family presence during cardiopulmonary resuscitation. N Engl J Med. 2013;368(11):1008–18.

- Madden E, Condon C. Emergency nurses' current practices and understanding of family presence during CPR. J Emerg Nurs. 2007;33(5):433–40.
- Twibell RS, Craig S, Siela D, Simmonds S, Thomas C. Being there: inpatients' perceptions of family presence during resuscitation and invasive cardiac procedures. Am J Crit Care. 2015;24(6):e108–15.
- Booth MG, Woolrich L, Kinsella J. Family witnessed resuscitation in UK emergency departments: a survey of practice. Eur J Anaesthesiol. 2004;21(9):725–8.
- Ong ME, Chan YH, Srither DE, Lim YH. Asian medical staff attitudes towards witnessed resuscitation. Resuscitation. 2004;60(1):45–50.
- 59. Lederman Z, Wacht O. Family presence during resuscitation: attitudes of Yale-New Haven Hospital staff. Yale J Biol Med. 2014;87(1):63–72.
- Redley B, Hood K. Staff attitudes towards family presence during resuscitation. Accid Emerg Nurs. 1996;4(3):145–51.
- Chalk A. Should relatives be present in the resuscitation room? Accid Emerg Nurs. 1995;3(2):58–61.
- Creamer KM. Family-witnessed resuscitation. Chest. 2003;124(2):769–70.
- Newton A. Witnessed resuscitation in critical care: the case against. Intensive Crit Care Nurs. 2002;18(3):146–50.
- 64. Macy C, Lampe E, O'Neil B, Swor R, Zalenski R, Compton S. The relationship between the hospital setting and perceptions of family-witnessed resuscitation in the emergency department. Resuscitation. 2006;70(1):74–9.
- Gomes BD, Dowd OP, Sethares KA. Attitudes of community hospital critical care nurses toward family-witnessed resuscitation. Am J Crit Care. 2019;28(2):142–8.
- Tudor K, Berger J, Polivka BJ, Chlebowy R, Thomas B. Nurses' perceptions of family presence during resuscitation. Am J Crit Care. 2014;23(6):e88–96.
- Offord RJ. Should relatives of patients with cardiac arrest be invited to be present during cardiopulmonary resuscitation? Intensive Crit Care Nurs. 1998;14(6):288–93.
- Van der Woning M. Should relatives be invited to witness a resuscitation attempt? A review of the literature. Accid Emerg Nurs. 1997;5(4):215–8.
- Belanger MA, Reed S. A rural community hospital's experience with family-witnessed resuscitation. J Emerg Nurs. 1997;23(3):238–9.
- Edwards L, Shaw DG. Care of the suddenly bereaved in cardiac care units: a review of the literature. Intensive Crit Care Nurs. 1998;14(3):144–52.
- Robinson SM, Mackenzie-Ross S, Campbell Hewson GL, Egleston CV, Prevost AT. Psychological effect of witnessed resuscitation on bereaved relatives. Lancet. 1998;352(9128):614–7.
- Fulbrook S. Legal implications of relatives witnessing resuscitation. Br J Theatr Nurs. 1998;7(10):33–5.
- Albarran J, Moule P, Benger J, McMahon-Parkes K, Lockyer L. Family witnessed resuscitation: the views and preferences of recently resuscitated hospital inpatients, compared to matched controls without the experience of resuscitation survival. Resuscitation. 2009;80(9):1070–3.
- 74. Oczkowski SJ, Mazzetti I, Cupido C, Fox-Robichaud AE, Canadian Critical Care S. Family presence during resuscitation: a Canadian Critical Care Society position paper. Can Respir J. 2015;22(4):201–5.
- Eichhorn DJ, Meyers TA, Mitchell TG, Guzzetta CE. Opening the doors: family presence during resuscitation. J Cardiovasc Nurs. 1996;10(4):59–70.

© Springer Nature Switzerland AG 2021 K. H. Todd et al. (eds.), Oncologic Emergency Medicine, https://doi.org/10.1007/978-3-030-67123-5_59

Caitlin Hodge and Brian Badgwell

Case Study

A 68-year-old woman presents to the emergency department with progressive nausea, vomiting, vague abdominal pain, and weight loss. She reports that her last bowel movement was 2 days ago and that she has lost 20 pounds in the past few months. On exam, she appears cachectic and jaundiced. Her vital signs are stable. Her abdomen is distended and tympanic. X-rays show dilated loops of bowel with air-fluid levels. A CT scan reveals a 5 cm pancreatic head mass and concern for liver metastases. Labs reveal a total bilirubin of 14 and a normal white count and lactate. She can climb two flights of stairs without difficulty.

IVs are placed and fluids are started. An NG tube is placed, which returns bilious output. Her abdominal discomfort improves after NG placement. She is admitted to the hospital for further work-up and resuscitation. An endoscopic ultrasound and ERCP are performed. Biopsies are taken, which confirm the diagnosis of pancreatic adenocarcinoma. Management options for her symptoms resulting from gastric outlet and biliary obstruction are discussed. After thorough discussion of risks and benefits, it is determined that her primary goals are to return home and to start chemotherapy. She then undergoes successful biliary stent placement followed by laparoscopic gastrojejunostomy. She recovers from these procedures and is discharged home.

C. Hodge

B. Badgwell (⊠)

Introduction

Palliative care, defined by the World Health Organization, is care to improve the quality of life of patients and their families facing the problems associated with life-threatening illness, through the prevention and relief of suffering [1]. Palliative care involves the assessment and treatment of pain and other physical, psychosocial, and spiritual problems. The first distinction that must be made in addressing palliative surgery is differentiating surgical palliative care from medical palliative care. Medical palliative care is the management of symptoms such as pain, nausea, cachexia, delirium, and fatigue. Surgical palliative care can broadly be separated into two main categories. The first is operative palliative surgical care in which surgical interventions are the treatment modality utilized to palliate patients with advanced or incurable illness. Palliative surgery is defined as surgery performed with the purpose of alleviating symptoms and improving quality of life [2]. This form of palliative surgical care is often encountered during surgical consultation or in the emergency department. The second form of palliative surgical care involves nonoperative care and decisions about the appropriate level of care in postoperative or trauma patients with life-threatening conditions or postoperative complications. This form of palliative surgical care is most often encountered in the intensive care unit or postoperative inpatient unit.

The focus of this chapter on palliative surgery will be clinical diagnoses evaluated for potential surgical intervention. Palliative surgical consultation is a frequent occurrence in hospitals that treat cancer patients. Approximately half of all inpatients undergoing surgical consultation at major cancer centers meet the criteria for palliative care. In a study at MD Anderson of over 1000 inpatient surgical consultations, 40% met the criteria for surgical palliative evaluation [3]. The low overall median survival of all patients (2.9 months) highlights the need for a selective approach to patients undergoing palliative surgical consultation. Attempts to identify variables associated with poor survival, and perhaps identify



Department of Surgery, Abington Hospital-Jefferson Health, Abington, PA, USA

Department of Surgical Oncology, Division of Surgery, MD Anderson Cancer Center, Houston, TX, USA e-mail: bbadgwell@mdanderson.org

patients that should be managed without surgery, were largely unsuccessful in this study although patients with two or more radiologic sites of disease and carcinomatosis had poorer survival. The risks of palliative surgery are significant as palliative surgical procedures have reported morbidity and mortality rates of 20–40% and 4–7%, respectively [3, 4]. In appropriately selected patients, palliative surgery can achieve up to 90% symptom resolution [4] and mean duration of symptom control of 135 days [5].

The most important factor in decision-making for surgical palliation is patient selection. While no single instrument has been defined as the best metric, several criteria have been proposed including symptom control, prognosis, preoperative performance, quality of life, tumor burden amenable to palliation, procedure-related morbidity and mortality, feasibility of nonsurgical options, anticipated duration of hospitalization, requirement for additional palliation, and cost [6]. Risk-benefit discussions during the consent process for palliative surgery are difficult owing to several limitations in existing data and research. First, the benefits of palliative surgery are largely unknown due to a paucity of high-quality, prospective, patient-reported outcome studies. Second, prognostication is difficult for advanced cancer patients, and the risks of surgery must be balanced against the estimated remaining length of life for patients undergoing palliative surgical consultation. There is no tool that specifically predicts complications for palliative intent surgical procedures. For example, the proportion of patients undergoing palliative surgical intervention for cancer within the American College of Surgeons National Surgical Quality Improvement (ACS-NSOIP) calculator pool is small, which limits the calculator's predictive ability for this population [7]. This calculator has been shown to significantly overestimate mortality but underestimate length of stay for palliative procedures [8]. Lastly, randomized clinical trials are difficult to perform in palliative populations and particularly in the palliative surgical population [9].

Despite a selective practice of surgical intervention, palliative surgery can still account for approximately 20% of a surgical oncologist's practice and over 1000 procedures per year at cancer centers [5, 10]. The frequency of palliative surgical consultation and intervention aids in the identification of common diagnoses and treatment patterns. Gastrointestinal obstruction is the most common indication for palliative surgical consultation at approximately 40% [3]. Gastrointestinal bleeding and wound complications/infections each account for 10% of palliative surgical consultations. Abdominal pain of unclear etiology is also common and often includes patients with constipation, ileus, carcinomatosis with resultant gastrointestinal dysfunction, and medication- or treatment-related side effects. The common theme throughout many of these palliative surgical consultations is the acute presentation of symptoms as these patients are

often evaluated in the hospital or emergency department. Although acute in nature, palliative surgical consultations rarely require urgent surgical intervention and allow for time to engage in multidisciplinary care discussions and a thorough evaluation of the associated risks and benefits.

Gastrointestinal Obstruction

Gastrointestinal obstruction is one of the most common indications for surgical consultation, even in patients without cancer. The standard approach by general surgeons in the evaluation of patients with obstruction includes a history to identify the anatomic site and degree of obstruction. The approach to assessment of gastrointestinal obstruction proceeds similarly in patients with and without cancer. Past surgical history is an important component of the subjective assessment for patients with obstruction as adhesions and hernias are the two most common causes of obstruction. Cancer is notably the third most common cause in the bowel obstruction differential diagnosis in patients undergoing general surgery evaluation. The objective assessment should first focus on vital signs to evaluate for hemodynamic instability. Fever and tachycardia are findings worrisome for ischemia. A complete physical examination should be performed. A focused assessment of the abdomen is necessary to evaluate for strangulated incisional or inguinal hernias, the degree of abdominal distention, and peritoneal signs. Laboratory analysis is necessary to manage electrolyte abnormalities and evaluate for leukocytosis, which could also indicate ischemia. Imaging is frequently obtained, first with plain films, but computed tomography imaging is often required in cancer patients not only to determine the site of obstruction but also to evaluate for sites of metastases within the abdomen, multifocal obstruction, and ascites. Treatment should begin during the evaluation of patients with bowel obstruction and may include nasogastric tube decompression, intravenous fluid resuscitation, and Foley catheter placement.

There are various definitions for gastrointestinal obstruction in cancer patients. Definitions include malignant bowel obstruction, patients with bowel obstruction secondary to recurrent cancer, patients with stage IV cancer and obstruction, and definitions based on anatomic site of obstruction [11–13]. A definition formulated during an international multi-institutional and multidisciplinary conference tasked with creating a definition for subsequent palliative trials defined malignant bowel obstruction as (1) clinical evidence of a bowel obstruction via history, physical exam, or radiographic examination, (2) bowel obstruction beyond the ligament of Treitz, (3) intra-abdominal primary cancer with incurable disease, or (4) non-intra-abdominal primary cancer with clear intraperitoneal disease [14]. A simplified definition for malignant bowel obstruction is blockage of the small or large intestine in a patient with advanced cancer [15]. As treatment varies depending on the anatomic site of obstruction, defining the obstruction as gastric outlet, small bowel, or large bowel can help identify differences in the utilization of endoscopic or surgical procedures and also identify differences in outcomes such as survival or symptom improvement [9, 13]. Regardless of the definition, bowel obstruction in patients with advanced cancer is common with reported rates of up to 42% of patients with advanced ovarian cancer and up to 24% of patients with advanced colorectal cancer developing obstruction during their lifetime [16]. Adding to the complexity of defining bowel obstruction in cancer patients is that somewhere between 3% and 40% of obstructions may have a benign etiology and are caused by adhesions or strictures not associated with malignancy [17].

Emesis and abdominal distention in patients with incurable cancer can be due to a myriad of factors other than mechanical obstruction. Patients with advanced cancer can suffer from electrolyte abnormalities, cachexia with metabolic derangements, pain medication side effects, constipation, autonomic dysfunction due to plexus involvement from malignancies with a tendency for perineural invasion, and chemotherapy/radiotherapy side effects. There are many surgical consultations for bowel obstruction that are ultimately found to have gastroparesis, ileus, or constipation.

Bowel obstruction in patients with advanced cancer is rarely a surgical emergency and typically allows time to consider multidisciplinary aspects of the patient's condition. Previous treatment, cancer stage, and prognosis are unique variables to consider in cancer patients with gastrointestinal obstruction. Although accurate prognostication is difficult, it is helpful to attempt to determine if a patient can recover from abdominal surgery and obtain a meaningful quality of life prior to death from their malignancy. The next aspect of surgical decision-making to consider is morbidity and mortality rates, which are considerable in patients undergoing palliative surgery. Morbidity and mortality rates are widely variable in the literature with a range of morbidity from 9% to 90% and a range of mortality from 9% to 40% [16]. A recent series from our institution demonstrated morbidity and mortality rates for surgical intervention of 44% and 5%, respectively [13].

Recognizing the lack of randomized trials and limited data from observational and retrospective studies of patients with advanced malignancy and bowel obstruction, various groups have attempted to provide consensus statements and treatment algorithms for these difficult clinical scenarios. The working group of the European Association for Palliative Care has provided clinical practice recommendations that state surgery should not be routinely undertaken and will only benefit selected patients with end-stage cancer and mechanical obstruction [16]. The working group went on to recommend absolute contraindications to surgery, such as previous abdominal surgery that showed diffuse metastatic cancer, involvement of the proximal stomach, and ascites that recurs rapidly after drainage. Relative contraindications include poor general performance status, poor nutritional status, and extra-abdominal metastases producing symptoms that are difficult to control. Many investigators have sought to identify variables associated with adverse outcomes. A systematic review of surgery for malignant bowel obstruction reported that poor performance status, diffuse carcinomatosis, previous radiotherapy, and small bowel site of obstruction were associated with failure of surgery [18]. Additionally, ascites and carcinomatosis are frequently reported as independent indicators of poor survival and also diminished ability to tolerate oral intake after palliative surgical intervention [19-21]. The combination of ascites and carcinomatosis creates a situation that is rarely palliated with surgery, other than venting gastrostomy tube placement.

Gastric Outlet Obstruction

Gastric outlet obstruction is defined as obstruction of the distal stomach or proximal duodenum and is most often associated with gastric, duodenal, or pancreatic malignancy. Figure 59.1 demonstrates a CT image of a patient with gastric outlet obstruction secondary to gastric cancer involving the pylorus. Gastric outlet obstruction accounts for only approximately 20% of palliative surgical consultations for gastrointestinal obstruction but has a low median survival of 3 months, highlighting the need for selective surgical intervention [9]. Management options include surgical loop gastrojejunostomy or endoscopic stenting. As the majority of



Fig. 59.1 CT image demonstrating gastric outlet obstruction

cancers that cause gastric outlet obstruction are associated with limited survival, prognostication plays an important role in treatment selection.

Although technically simple, morbidity and mortality rates after loop gastrojejunostomy are significant, and temporary delayed gastric emptying can occur [17]. Laparoscopic gastrojejunostomy may carry less morbidity and is associated with a lower incidence of delayed gastric emptying and faster resumption of oral intake [22]. Endoscopic stents, on the other hand, have less risk of morbidity and mortality but lack the durability of a surgical bypass [18, 19]. Stents can be complicated by occlusion or migration and are more likely to require re-intervention [23, 24]. A Cochrane review found that despite similar rates of technical success, stents were associated with a slightly shorter time to oral intake and a shorter length of stay [24]. This allows for earlier resumption of chemotherapy, although a survival difference has not been demonstrated [25]. Furthermore, stents are associated with a lower overall cost for index hospitalization and lower rates of 90-day readmission [23]. However, given the rates of recurrent obstructive symptoms and re-interventions in patients undergoing endoscopic stent placement, several authors have proposed surgery should be considered in patients with a life expectancy of 2 months or longer [21, 22]. A new technique for treatment of gastric outlet obstruction is endoscopic ultrasound-guided gastroenterostomy, which has shown similar technical and clinical success with less need for reintervention impaired to endoscopic stenting [26], but longterm follow-up studies are needed [27].

Venting gastrostomy tubes are another option for patients with symptoms of nausea and emesis who have contraindications to surgical or endoscopic palliation. Gastrostomy tube placement may be performed through open surgery, laparoscopic surgery, endoscopy, or interventional radiology, although preference is given to the least invasive procedure possible as median post procedure survival rates are reported as only a few weeks [28-30]. Both endoscopic and interventional radiologic placement can be complicated by tube migration, leakage, and infections of the tube site, but literature reviews suggest endoscopic placement has fewer complications [24]. A retrospective review found that for patients who received a decompressing gastrostomy tube, a longer interval to placement was not associated with a change in overall survival, which suggests that venting gastrostomy tubes are often placed late in the course of disease [30].

Small Bowel Obstruction

Small bowel obstruction is defined as obstruction from the terminal portion of the duodenum to the ileocecal valve.

Small bowel obstruction represents the most common indication for palliative surgical consultation in patients with gastrointestinal obstruction (64%) but has a similar median survival (3.5 months) to gastric outlet obstruction [13]. Only 25% of patients with small bowel obstruction undergo surgical intervention with the majority (52%) undergoing nonoperative/nonprocedural management and 24% undergoing endoscopic/interventional procedures. The majority of endoscopic or interventional radiologic procedures in this population are venting gastrostomy tubes, as stents are typically not an option for small bowel obstructions. Many patients

lation are venting gastrostomy tubes, as stents are typically not an option for small bowel obstructions. Many patients with advanced cancer and bowel obstruction have had previous surgery, adding difficulty in the differentiation of an obstruction due to malignancy from an obstruction due to benign adhesive disease. The surgical procedure is often not decided upon until completing exploration of the abdomen, and the two most common approaches are either bowel resection or intestinal bypass. A venting gastrostomy tube can be placed at surgery for patients with disease prohibiting resection or bypass, and conditions felt to be at high risk for early re-obstruction.

Large Bowel Obstruction

As many as 8-29% of patients with primary colorectal cancer present with malignant colonic obstruction [31]. Management options include surgical options of bypass, bowel resection or diverting ostomy placement, endoscopic stenting, or supportive care. Stents can be used as a palliative therapy in patients who are not good operative candidates or can be used as a bridge to surgery in patients presenting with an obstruction. A retrospective review found that stenting was clinically successful in 86% of patients undergoing palliative stenting and 89% of patients undergoing stenting as a bridge to surgery [32]. Stents are associated with a low risk of perforation [32] but can migrate or obstruct and thus require re-intervention [33] or urgent surgery in up to one third of patients [34]. Peritoneal carcinomatosis is associated with decreased technical and clinical success in stent placement [35], although a lower likelihood of delayed perforation [36]. The advantages of stent placement include shorter length of stay, avoidance of a stoma, and allowance for a better oncologic resection in patients who are later amenable to curative surgery [31, 33, 34]. Patients with inoperable malignant bowel obstruction can be treated with medical therapy including antiemetics, steroids, anti-secretory agents, octreotide, and hyoscine butylbromide [37]. One retrospective study found that 49% of these obstructions resolved with medical management, with functional mechanisms of obstruction more likely to resolve [37].

Gastrointestinal Bleeding

Similar to bowel obstruction, gastrointestinal bleeding in patients with cancer can be due to benign or malignant causes. The initial evaluation focuses on determining the severity of bleeding and assessing for hemodynamic instability. Utilizing the same initial approach taught through advanced trauma and life support training, surgeons assess the airway, breathing, and circulation of patients with gastrointestinal bleeding while ensuring adequate intravenous access. During this time, laboratory analysis should begin to include a complete blood count, blood typing and crossmatching, coagulation factors, and electrolytes with BUN and creatinine. A nasogastric tube and Foley catheter are often required for patients with active bleeding. Most patients are hemodynamically stable which allows time for medical management and diagnostic workup. Endoscopy is the primary modality utilized in gastrointestinal bleeding for diagnosis and therapy. Other less commonly required tests include tagged red blood cell scans, arteriography, and capsule endoscopy. As patients with advanced cancer are often best treated without surgical intervention, therapeutic options frequently involve embolization performed by interventional radiologists.

Bleeding secondary to tumor or treatment-related complications can involve many site-specific diagnostic and treatment issues. Tumors of the gastroesophageal junction and stomach often account for anemia through a slow rate of gastrointestinal hemorrhage and rarely require urgent surgery. In a recent review of 289 patients with advanced gastric cancer from Massachusetts General Hospital, only 3.5% required emergent surgery at presentation, of which none were performed for bleeding [38]. In addition, 233 patients in this series were managed without resection of the primary tumor, of which only 6 patients required subsequent emergency surgery for obstruction or perforation and no patient required surgery for bleeding. The low rate of surgical intervention for gastric hemorrhage is likely due to the many options available for bleeding secondary to tumor involvement such as endoscopic interventions, chemotherapy, and radiotherapy. For patients with gastric cancer, palliative radiotherapy can control bleeding in 70% of patients with a low rate of requirement for additional interventions during the patient's remaining life [39] and a duration of symptom control of 15 weeks [40].

Bleeding from small bowel tumors can be secondary to primary or metastatic malignancy. The small intestine can be a difficult diagnostic challenge as endoscopy will typically only assess to the level of the duodenum from above and terminal ileum from below. The remainder of the small bowel may involve arteriography or capsule endoscopy to accurately localize the site of bleeding. Bleeding from the large intestine and rectum can often be localized with colonoscopy, with the majority of cases attributable to primary colon and rectal cancer. Primary tumor response rates to palliative chemotherapy and radiation (for rectal cancer) are good and infrequently require emergent surgery.

Wound Problems and Infections

Palliative wound care can be a challenging clinical scenario that is of extreme importance to the patient. Problems related to wounds include bleeding, exudate, odor, pain, and limitations in function. Wound complications and infections are a frequent indication for palliative surgical consultation, representing 10% of palliative surgical inpatient consults at comprehensive cancer centers [3]. Treatment approaches include local wound care, excision, amputation, systemic therapy, and radiation. Traditional wound care management strategies are limited by the impaired healing of patients with advanced malignancy or patients that have received recent immunosuppressive therapy. Malignant fungating wounds are a unique challenge due to malodor and discharge that can affect up to 5% of people with cancer [41]. Figure 59.2 demonstrates a malignant fungating squamous cell cancer of the posterior scalp. Systematic reviews have identified few highquality studies or effective therapies to guide the topical treatment of malodor and discharge in fungating wounds. However, a Cochrane review found one study showing that foam castings containing silver provided improved control of malodor, and another study showed similar results with regard to exudate, malodor, and pain with manuka honey and nanocrystalline silver dressings [42]. Although typically



Fig. 59.2 Fungating malignant wound of the posterior scalp

used for more short-term wound care, one small study found that the use of negative pressure wound therapy for malignant wounds led to decreased pain, fewer dressing changes, and increased resumption of social interactions, with an average duration of 49 days [43]. Surgery is rarely an option but may be a last resort in patients that have wounds refractory to other therapy and are appropriate surgical candidates. Involvement of plastic surgery for advanced wound closure techniques is frequently required in patients undergoing surgery for wound problems.

Obstructive Jaundice

Options for treating biliary obstruction in patients with cancer include endoscopic stent placement, percutaneous catheter placement, or surgical bypass. Prognostication, although difficult and often inaccurate, can help identify the optimal treatment method. Endoscopic stent placement via endoscopic retrograde cholangiopancreatography (ERCP) is now the gold standard for biliary drainage as it is safe and effective [44]. Recurrent obstruction can occur but may be decreased with the use of self-expanding metallic stents. A systematic review and meta-analysis showed lower occlusion rates but increased rates of migration and pancreatitis for covered stents compared to uncovered stents, with no significant difference in survival or overall complication rates [45]. Percutaneous catheter placement by interventional radiology is often reserved for patients that fail endoscopic stent placement, as percutaneous procedures are more invasive. A systematic review and meta-analysis found that percutaneous transhepatic biliary drainage has a higher incidence of bleeding and dislocation but lower incidence of cholangitis and pancreatitis compared to endoscopic stenting. Yet, they found no difference in therapeutic success rate, overall complication rate, or 30-day mortality [46]. Endoscopic ultrasound-guided biliary drainage is a more recent procedure that can be used if ERCP stent placement is not feasible [47]. Several randomized controlled trials have found similar stent patency, adverse events, and re-interventions in centers where this technique is available [48].

Surgery is required in a fraction of patients but should be considered in patients that fail endoscopic attempts, live far away from referral centers with endoscopic expertise, or have other indications for surgical intervention such as concomitant gastric outlet obstruction. Even in the setting of gastric outlet obstruction, a gastric bypass and biliary stent may be preferred to a double bypass, as a double bypass procedure is associated with a higher rate of both biliary and non-biliary complications [49]. Figure 59.3 shows a CT image of a patient with concomitant bile duct obstruction and duodenal narrowing secondary to a locally invasive pancreatic neuroendocrine tumor. *Roux-en-Y* hepaticojejunostomy/choledo-



Fig. 59.3 CT image demonstrating a locally invasive pancreatic cancer causing narrowing of the duodenum (*black arrow*) and bile duct obstruction requiring metallic stent placement (*white arrow*)

chojejunostomy is the most frequently performed method of palliation and involves anastomosis of a 40–60 cm Roux limb to the bile duct in an end-to-side or side-to-side fashion. Anastomosis of the bowel to the gallbladder (cholecystojejunostomy) is a simpler method of biliary bypass that may be conducive to a laparoscopic approach, with acceptable results in small series [50]. However, a large populationbased analysis of the SEER-Medicare data found a biliary intervention rate of 7.5% for patients undergoing cholecystojejunostomy and only 2.9% for patients undergoing bile duct bypass [51].

Bowel Perforation

As with gastrointestinal bleeding and obstruction, there are benign and tumor-related causes of bowel perforation. Common benign causes of perforation include peptic ulcer disease and diverticulitis. Tumor-related causes include direct tumor invasion with perforation but also the side effects of cancer treatment such as immunosuppression, radiation effects, steroid administration, and the direct effects of chemotherapy that may render cancer patients more prone to bowel perforation. Cancer patients may also be more prone to iatrogenic bowel perforation due to the need for frequent endoscopic and interventional radiology procedures. As many cancer patients with bowel perforation have advanced or incurable disease, the optimal treatment is often based on a balance between the clinical presentation and oncologic prognosis. The majority of cancer patients with bowel perforation are treated with surgery with a 30-day mortality rate of 15% and morbidity rate of 46% [52].

Nonoperative care is another option, particularly in patients with incurable disease, and may have similar outcomes in select patients without abdominal tenderness, limited extent of perforation or contained free air, and aggressive treatment with antibiotics and drain placement [52]. One retrospective review found better outcomes after palliative surgery for intestinal perforation in patients who had higher albumin, low ECOG performance status, and absence of dyspnea and thus recommended consideration of nonoperative management for patients with unfavorable prognostic factors [53].

A unique and complex clinical situation is bowel perforation or fistula formation during treatment with bevacizumab. Bevacizumab is a humanized monoclonal antibody to vascular endothelial growth factor that has been proven efficacious in a number of disease sites, including several phase III randomized trials of patients with metastatic colon cancer [54]. Although generally well tolerated, bevacizumab has been associated with gastrointestinal perforation in 1-3.5% of patients [55, 56]. In a large retrospective review, risk factors for perforation included poorly differentiated signet ring cell carcinoma, stent use, and rectal location of primary and intact primary, although complications requiring surgery did not change survival [56]. Wound healing complications are increased in patients that undergo surgery during treatment. and most surgeons exercise caution in performing a gastrointestinal anastomosis in patients with bevacizumab-associated perforation [55].

Anorectal Infections

Anorectal infections and abscesses in noncancer patients are typically straightforward in management consisting of incision and drainage with complex treatment required in only a minority of patients. Anorectal infections in patients with immunosuppression, neutropenia, recent chemotherapy, or stem cell transplant and advanced hematologic disease can be a difficult palliative situation with considerable treatment and quality of life implications. Older reports of anorectal disease in neutropenic patients detailed associated mortality rates of up to 50%, while more recent reports suggest improvements in survival [57]. Anorectal infections in cancer patients are classified as either an abscess or perianal infectious process (pain or erythema without abscess/fluid). Necrotizing soft tissue infections are rare ($\sim 2\%$) but associated with significant mortality [57]. Patients often need imaging to evaluate for abscess or fluid formation, as physical exam findings can be misleading. Patients with fluid typically undergo an exam under anesthesia with drainage and seton placement or catheter placement via interventional radiology. Patients with a perianal infectious process without documentation of fluid are managed with antibiotics and close monitoring.

Ascites

Malignant ascites has a detrimental effect on quality of life with associated symptoms of abdominal pain, dyspnea, nausea, vomiting, anorexia, impaired movement, and fatigue [58, 59]. It is associated with limited survival, on the order of 1–6 months [58, 60–63], with only 11% of patients surviving greater than 6 months after the development of malignant ascites [64]. However, ovarian cancer has a more favorable prognosis than other primaries in the setting of ascites with reported survival rates of 24 months [64]. There are a wide variety of treatment options including diuretic administration, fluid restriction, systemic chemotherapy, intermittent paracentesis, peritoneal drainage catheters, and hyperthermic intraperitoneal chemotherapy.

Paracentesis is the most common treatment modality. In a physician practice survey, 98% of practitioners reported using paracentesis, and 89% reported it to be effective [65]. While paracentesis provides temporary relief for 90% of patients [66], repeat large-volume paracentesis is associated with decreased quality of life as paracentesis is only performed once symptoms recur [67]. Tunneled peritoneal catheters, such as the PleurX system, can provide a more durable option by allowing more frequent and smaller-volume drainage that can be performed outside of the hospital [68-71]. Tunneled catheters can be placed under fluoroscopic, ultrasound, or CT guidance and have reported technical success rates of close to a hundred percent in several retrospective and nonrandomized prospective studies [68-73]. The majority of patients with the PleurX catheter report good control of their symptoms, and the need for further interventions to restore catheter function is infrequent [71]. Surgeons infrequently perform peritoneal venous shunt placement due to its less effective control of ascites as well as higher complication rates, including sepsis, heart failure, disseminated intravascular coagulation, and shunt malfunction or infection [67].

Hyperthermic intraperitoneal chemotherapy (HIPEC) is combined with cytoreductive surgery in the treatment of peritoneal surface malignancy, appendiceal mucinous neoplasms, and colorectal/ovarian carcinomatosis. However, cytoreduction combined with HIPEC is a morbid procedure with an established mortality rate and lengthy post procedure hospitalization that is not generally appropriate for palliative surgical scenarios. Laparoscopic HIPEC, without cytoreduction, has been performed in patients with malignant ascites with excellent results in small series (Fig. 59.4). The laparoscopic approach appears to alleviate much of the morbidity a multi-institutional analysis of 52 patients undergoing laparoscopic HIPEC reported a complication rate of only 6% with no postoperative mortalities [74]. Remarkably, the laparoscopic HIPEC procedure prevented re-accumulation of ascites in all but one patient [74]. Furthermore, small studies have found an average increase in Karnofsky performance scores of 20 points after treatment [74, 75].



Fig. 59.4 Intraoperative cannula placement for laparoscopic hyperthermic intraperitoneal chemotherapy administration

Abdominal Pain in Unique Patient Populations

Celiac Plexus Involvement

Tumors with a propensity for perineural invasion, such as pancreatic cancer, can cause debilitating abdominal pain that radiates to the back. Pain in cancer patients, however, is infrequently attributable to a single cause but is more often a multi-factorial syndrome of tumor-related causes, treatment-related causes, and chronic pre-existing pain unrelated to cancer or its treatment. Opioids are the first line of treatment and often the only treatment that is needed. Palliative radiation is another treatment option for patients with pain secondary to celiac plexus involvement [76]. Celiac plexus neurolysis is a good local treatment option that can be performed through a percutaneous or endoscopic approach. Recent systematic reviews of plexus block procedures demonstrate improvements in pain with side effects usually limited to diarrhea, hypotension, and temporary increased levels of pain [77]. With multiple nonoperative options to alleviate celiac plexus-associated pain,

surgical chemical blocks are relegated to the intraoperative scenario of finding unresectable disease during attempted pancreaticoduodenectomy. In such a situation, there is evidence from a randomized clinical trial that an intraoperative celiac block can lower pain in patients with preoperative pain and also prevent pain in patients without preoperative pain [78]. In situations where unresectable pancreatic cancer is detected during diagnostic laparoscopy, a laparoscopic celiac block has similarly been proven efficacious in reducing pain scores [79].

Neutropenia

Abdominal pain in neutropenic cancer patients presents a palliative clinical challenge as 90-day mortality rates are approximately 50%. The differential diagnosis includes causes of abdominal pain common in general surgery consultation such as bowel obstruction, diverticulitis, and appendicitis but also cancer treatment-related causes such as neutropenic enterocolitis and *Clostridium difficile* colitis [80]. Mortality associated with surgical intervention in the presence of neutropenia has been reported as high as 57% [81]. Surgeons will often deliberately delay treatment to allow for resolution of neutropenia, if possible [80].

Multiple Myeloma

Multiple myeloma is a plasma cell malignancy with systemic overproduction of antibodies resulting in bone pain, anemia, and renal insufficiency, with progressive tumor formation resulting in bone marrow failure. Surgical interventions are primarily palliative for long bone compression fractures or spinal cord compression. Abdominal pain requiring surgical consultation in myeloma patients is a serious condition with an associated 90-day mortality rate of 43% [82]. The differential diagnosis in this unique patient population is notable for the frequency of neutropenic enterocolitis (22%) and bowel perforation (13%) [82]. Prompt attention should be given to new complaints of abdominal pain in patients with myeloma as surgery may be required, although consideration should be given to the frequent comorbid conditions, severe sepsis, and recent administration of chemotherapy in the setting of a disease with often limited survival.

Outcome Measures

The palliative surgical literature is difficult to interpret due to the lack of commonly accepted outcome measures. In a review of studies from the palliative surgical literature, quality of life measurements were only included in 17% of studies, while morbidity and mortality were reported in 61% [83]. Morbidity and mortality, although important outcome measures in the risk analysis of surgery, must be balanced against patient-reported benefits of palliative surgery. Adding to the complexity of palliative surgical studies are the high attrition rate and difficulty in administering burdensome general quality of life instruments [84]. Observational outcome measures may provide some improvement in rates of postoperative symptom evaluation [85]. Future studies of easily administered, quick, patient-reported outcomes will be needed to identify the optimal outcome measure and then identify variables associated with outcome to select patients appropriately for surgery [86].

Conclusion

In summary, palliative surgical care is becoming increasingly recognized as critical in the multidisciplinary treatment of cancer patients. Traditional literature has focused on outcomes of morbidity and mortality with recent efforts to identify optimal patient-reported outcomes. There are few standardized guidelines for surgical intervention, and the decision to proceed with surgery is based on patient, family, and provider discussions incorporating a risk versus benefit model. Careful consideration is given to prognosis, although difficult and sometimes inaccurate, as well as future oncologic treatment options. As patients undergoing palliative surgical consultation are often dealing with severe symptoms, these discussions should be started as soon as possible, and attempts to anticipate future palliative clinical scenarios are helpful.

References

- World Health Organization. Palliative care program, http://www. who.int/cancer/palliative/en/ http://www.who.int/cancer/palliative/ en/
- McCahill LE, Dunn GP, Mosenthal AC, Milch RA, Krouse RS. Palliation as a core surgical principle: part 1. J Am Coll Surg. 2004;199(1):149–60.
- Badgwell BD, Smith K, Liu P, Bruera E, Curley SA, Cormier JN. Indicators of surgery and survival in oncology inpatients requiring surgical evaluation for palliation. Support Care Cancer. 2009;17(6):727–34.
- 4. Miner TJ, Cohen J, Charpentier K, McPhillips J, Marvell L, Cioffi WG. The palliative triangle: improved patient selection and outcomes associated with palliative operations. Arch Surg. 2011;146(5):517–22.
- Miner TJ, Brennan MF, Jaques DP. A prospective, symptom related, outcomes analysis of 1022 palliative procedures for advanced cancer. Ann Surg. 2004;240(4):719–26; discussion 26–7.
- Foster D, Shaikh MF, Gleeson E, Babcock BD, Ringold D, Bowne WB. Palliative surgery for advanced cancer: identifying evidencebased criteria for patient selection: case report and review of literature. J Palliat Med. 2016;19(1):22–9.

- Vidri RJ, Blakely AM, Kulkarni SS, Vaghjiani RG, Heffernan DS, Harrington DT, et al. American College of Surgeons National Surgical Quality Improvement Program as a qualitymeasurement tool for advanced cancer patients. Ann Palliat Med. 2015;4(4):200–6.
- Rodriguez RA, Fahy BN, McClain M, Morris KT. Estimation of risk in cancer patients undergoing palliative procedures by the American college of surgeons risk calculator. J Palliat Med. 2016;19(10):1039–42.
- Whalen GF, Kutner J, Byock I, Gerard D, Stovall E, Sieverding P, et al. Implementing palliative care studies. J Pain Symptom Manag. 2007;34(1 Suppl):S40–8.
- McCahill LE, Krouse R, Chu D, Juarez G, Uman GC, Ferrell B, et al. Indications and use of palliative surgery-results of Society of Surgical Oncology survey. Ann Surg Oncol. 2002;9(1):104–12.
- Abbas SM, Merrie AE. Resection of peritoneal metastases causing malignant small bowel obstruction. World J Surg Oncol. 2007;5:122.
- Francescutti V, Miller A, Satchidanand Y, Alvarez-Perez A, Dunn KB. Management of bowel obstruction in patients with stage IV cancer: predictors of outcome after surgery. Ann Surg Oncol. 2013;20(3):707–14.
- Badgwell BD, Contreras C, Askew R, Krouse R, Feig B, Cormier JN. Radiographic and clinical factors associated with improved outcomes in advanced cancer patients with bowel obstruction. J Palliat Med. 2011;14(9):990–6.
- Anthony T, Baron T, Mercadante S, Green S, Chi D, Cunningham J, et al. Report of the clinical protocol committee: development of randomized trials for malignant bowel obstruction. J Pain Symptom Manag. 2007;34(1 Suppl):S49–59.
- Badgwell B. Oncologic emergencies. In: Feig BW, Ching DC, editors. The MD Anderson surgical oncology handbook. 5th ed. Philadelphia: Lippincott Williams & Wilkins; 2011. p. 764–77.
- Ripamonti C, Twycross R, Baines M, Bozzetti F, Capri S, De Conno F, et al. Clinical-practice recommendations for the management of bowel obstruction in patients with end-stage cancer. Support Care Cancer. 2001;9(4):223–33.
- Tang E, Davis J, Silberman H. Bowel obstruction in cancer patients. Arch Surg. 1995;130(8):832–6; discussion 6–7.
- Santangelo ML, Grifasi C, Criscitiello C, Giuliano M, Calogero A, Dodaro C, et al. Bowel obstruction and peritoneal carcinomatosis in the elderly. A systematic review. Aging Clin Exp Res. 2017;29(Suppl 1):73–8.
- Blair SL, Chu DZ, Schwarz RE. Outcome of palliative operations for malignant bowel obstruction in patients with peritoneal carcinomatosis from nongynecological cancer. Ann Surg Oncol. 2001;8(8):632–7.
- Dalal KM, Gollub MJ, Miner TJ, Wong WD, Gerdes H, Schattner MA, et al. Management of patients with malignant bowel obstruction and stage IV colorectal cancer. J Palliat Med. 2011;14(7):822–8.
- Amikura K, Sakamoto H, Yatsuoka T, Kawashima Y, Nishimura Y, Tanaka Y. Surgical management for a malignant bowel obstruction with recurrent gastrointestinal carcinoma. J Surg Oncol. 2010;101(3):228–32.
- Ojima T, Nakamori M, Nakamura M, Katsuda M, Hayata K, Yamaue H. Laparoscopic gastrojejunostomy for patients with unresectable gastric cancer with gastric outlet obstruction. J Gastrointest Surg. 2017;21(8):1220–5.
- Fisher AV, Hanlon B, Fernandes-Taylor S, Schumacher JR, Lawson EH, Ronnekleiv-Kelly SM, et al. Natural history and cost analysis of surgical bypass versus endoscopic stenting for the palliative management of malignant gastric outlet obstruction. HPB (Oxford). 2020;22(4):529–36.
- Upchurch E, Ragusa M, Cirocchi R. Stent placement versus surgical palliation for adults with malignant gastric outlet obstruction. Cochrane Database Syst Rev. 2018;5(5):CD012506.

- Uemura S, Iwashita T, Iwata K, Mukai T, Osada S, Sekino T, et al. Endoscopic duodenal stent versus surgical gastrojejunostomy for gastric outlet obstruction in patients with advanced pancreatic cancer. Pancreatology. 2018; https://doi.org/10.1016/j. pan.2018.04.015.
- 26. Chen YI, Itoi T, Baron TH, Nieto J, Haito-Chavez Y, Grimm IS, et al. EUS-guided gastroenterostomy is comparable to enteral stenting with fewer re-interventions in malignant gastric outlet obstruction. Surg Endosc. 2017;31(7):2946–52.
- Tringali A, Giannetti A, Adler DG. Endoscopic management of gastric outlet obstruction disease. Ann Gastroenterol. 2019;32(4):330–7.
- 28. Laval G, Arvieux C, Stefani L, Villard ML, Mestrallet JP, Cardin N. Protocol for the treatment of malignant inoperable bowel obstruction: a prospective study of 80 cases at Grenoble University Hospital Center. J Pain Symptom Manag. 2006;31(6):502–12.
- Brooksbank MA, Game PA, Ashby MA. Palliative venting gastrostomy in malignant intestinal obstruction. Palliat Med. 2002;16(6):520–6.
- Pinard KA, Goring TN, Egan BC, Koo DJ. Drainage Percutaneous Endoscopic Gastrostomy for malignant bowel obstruction in gastrointestinal cancers: prognosis and implications for timing of palliative intervention. J Palliat Med. 2017;20(7):774–8.
- 31. Ribeiro IB, Bernardo WM, Martins BDC, de Moura DTH, Baba ER, Josino IR, et al. Colonic stent versus emergency surgery as treatment of malignant colonic obstruction in the palliative setting: a systematic review and meta-analysis. Endosc Int Open. 2018;6(5):E558–e67.
- 32. Gleditsch D, Søreide OK, Nesbakken A. Managing malignant colorectal obstruction with self-expanding stents. A closer look at bowel perforations and failed procedures. J Gastrointest Surg. 2016;20(9):1643–9.
- 33. Atukorale YN, Church JL, Hoggan BL, Lambert RS, Gurgacz SL, Goodall S, et al. Self-expanding metallic stents for the management of emergency malignant large bowel obstruction: a systematic review. J Gastrointest Surg. 2016;20(2):455–62.
- 34. Genzor Ríos SJ, González-Nicolás Trebol MT, Aguirre Prat N, Cantalejo Díaz M, Rodríguez Artigas JM, Medrano Peña J, et al. Stent performance in obstructive left colon cancer. Unicentric and cross-sectional study. Cir Esp. 2019;97(7):397–404.
- Park JJ, Rhee K, Yoon JY, Park SJ, Kim JH, Kim JH, et al. Impact of peritoneal carcinomatosis on clinical outcomes of patients receiving self-expandable metal stents for malignant colorectal obstruction. Endoscopy. 2018;50(12):1163–74.
- 36. Lee YJ, Yoon JY, Park JJ, Park SJ, Kim JH, Youn YH, et al. Clinical outcomes and factors related to colonic perforations in patients receiving self-expandable metal stent insertion for malignant colorectal obstruction. Gastrointest Endosc. 2018;87(6): 1548–57, e1.
- 37. Romeo M, de Los LGM, Cuadra Urteaga JL, Vilà L, Ahlal S, Indacochea A, et al. Outcome prognostic factors in inoperable malignant bowel obstruction. Support Care Cancer. 2016;24(11):4577–86.
- Schmidt B, Look-Hong N, Maduekwe UN, Chang K, Hong TS, Kwak EL, et al. Noncurative gastrectomy for gastric adenocarcinoma should only be performed in highly selected patients. Ann Surg Oncol. 2013;20(11):3512–8.
- 39. Kim MM, Rana V, Janjan NA, Das P, Phan AT, Delclos ME, et al. Clinical benefit of palliative radiation therapy in advanced gastric cancer. Acta Oncol. 2008;47(3):421–7.
- Lee YH, Lee JW, Jang HS. Palliative external beam radiotherapy for the treatment of tumor bleeding in inoperable advanced gastric cancer. BMC Cancer. 2017;17(1):541.
- Grocott P, Gethin G, Probst S. Malignant wound management in advanced illness: new insights. Curr Opin Support Palliat Care. 2013;7(1):101–5.

- Adderley UJ, Holt IG. Topical agents and dressings for fungating wounds. Cochrane Database Syst Rev. 2014;2014(5):CD003948.
- 43. Riot S, de Bonnecaze G, Garrido I, Ferron G, Grolleau JL, Chaput B. Is the use of negative pressure wound therapy for a malignant wound legitimate in a palliative context? "The concept of NPWT ad vitam": a case series. Palliat Med. 2015;29(5):470–3.
- 44. Perinel J, Adham M. Palliative therapy in pancreatic cancerpalliative surgery. Transl Gastroenterol Hepatol. 2019;4:28.
- 45. Moole H, Bechtold ML, Cashman M, Volmar FH, Dhillon S, Forcione D, et al. Covered versus uncovered self-expandable metal stents for malignant biliary strictures: a meta-analysis and systematic review. Indian J Gastroenterol. 2016;35(5):323–30.
- 46. Duan F, Cui L, Bai Y, Li X, Yan J, Liu X. Comparison of efficacy and complications of endoscopic and percutaneous biliary drainage in malignant obstructive jaundice: a systematic review and metaanalysis. Cancer Imaging. 2017;17(1):27.
- Park DH. Endoscopic ultrasound-guided biliary drainage of hilar biliary obstruction. J Hepatobiliary Pancreat Sci. 2015;22(9):664–8.
- Nabi Z, Reddy DN. Endoscopic palliation for biliary and pancreatic malignancies: recent advances. Clin Endosc. 2019;52(3):226–34.
- 49. Angelico R, Khan S, Dasari B, Marudanayagam R, Sutcliffe RP, Muiesan P, et al. Is routine hepaticojejunostomy at the time of unplanned surgical bypass required in the era of self-expanding metal stents? HPB (Oxford). 2017;19(4):365–70.
- Toumi Z, Aljarabah M, Ammori BJ. Role of the laparoscopic approach to biliary bypass for benign and malignant biliary diseases: a systematic review. Surg Endosc. 2011;25(7):2105–16.
- Urbach DR, Bell CM, Swanstrom LL, Hansen PD. Cohort study of surgical bypass to the gallbladder or bile duct for the palliation of jaundice due to pancreatic cancer. Ann Surg. 2003;237(1):86–93.
- 52. Badgwell B, Feig BW, Ross MI, Mansfield PF, Wen S, Chang GJ. Pneumoperitoneum in the cancer patient. Ann Surg Oncol. 2007;14(11):3141–7.
- 53. Ng DW, Teo MC, Yee AC, Ng KH, Wong TH. Surgery for the palliation of intestinal perforation from secondary metastases in advanced malignancies. Asia Pac J Clin Oncol. 2016;12(4):453–9.
- 54. Badgwell BD, Camp ER, Feig B, Wolff RA, Eng C, Ellis LM, et al. Management of bevacizumab-associated bowel perforation: a case series and review of the literature. Ann Oncol. 2008;19(3):577–82.
- 55. Scappaticci FA, Fehrenbacher L, Cartwright T, Hainsworth JD, Heim W, Berlin J, et al. Surgical wound healing complications in metastatic colorectal cancer patients treated with bevacizumab. J Surg Oncol. 2005;91(3):173–80.
- 56. Bong JW, Lee JL, Kim CW, Yoon YS, Park IJ, Lim SB, et al. Risk factors and adequate management for complications of bevacizumab treatment requiring surgical intervention in patients with metastatic colorectal cancer. Clin Colorectal Cancer. 2018;17(4):e639–e45.
- Badgwell BD, Chang GJ, Rodriguez-Bigas MA, Smith K, Lupo PJ, Frankowski RF, et al. Management and outcomes of anorectal infection in the cancer patient. Ann Surg Oncol. 2009;16(10):2752–8.
- Sangisetty SL, Miner TJ. Malignant ascites: a review of prognostic factors, pathophysiology and therapeutic measures. World J Gastrointest Surg. 2012;4(4):87–95.
- 59. Chung M, Kozuch P. Treatment of malignant ascites. Curr Treat Options in Oncol. 2008;9(2–3):215–33.
- Adam RA, Adam YG. Malignant ascites: past, present, and future. J Am Coll Surg. 2004;198(6):999–1011.
- Maeda H, Kobayashi M, Sakamoto J. Evaluation and treatment of malignant ascites secondary to gastric cancer. World J Gastroenterol. 2015;21(39):10936–47.
- Garrison RN, Kaelin LD, Galloway RH, Heuser LS. Malignant ascites. Clinical and experimental observations. Ann Surg. 1986;203(6):644–51.
- Mackey JR, Venner PM. Malignant ascites: demographics, therapeutic efficacy and predictors of survival. Can J Oncol. 1996;6(2):474–80.

- Stukan M. Drainage of malignant ascites: patient selection and perspectives. Cancer Manag Res. 2017;9:115–30.
- Lee CW, Bociek G, Faught W. A survey of practice in management of malignant ascites. J Pain Symptom Manag. 1998;16(2):96–101.
- Rosenberg SM. Palliation of malignant ascites. Gastroenterol Clin N Am. 2006;35(1):189–99, xi.
- Hodge C, Badgwell BD. Palliation of malignant ascites. J Surg Oncol. 2019;120(1):67–73.
- 68. Knight JA, Thompson SM, Fleming CJ, Bendel EC, Neisen MJ, Neidert NB, et al. Safety and effectiveness of palliative tunneled peritoneal drainage catheters in the management of refractory malignant and non-malignant ascites. Cardiovasc Intervent Radiol. 2018;41(5):753–61.
- Narayanan G, Pezeshkmehr A, Venkat S, Guerrero G, Barbery K. Safety and efficacy of the PleurX catheter for the treatment of malignant ascites. J Palliat Med. 2014;17(8):906–12.
- Wong BC, Cake L, Kachuik L, Amjadi K. Indwelling peritoneal catheters for managing malignancy-associated ascites. J Palliat Care. 2015;31(4):243–9.
- Courtney A, Nemcek AA Jr, Rosenberg S, Tutton S, Darcy M, Gordon G. Prospective evaluation of the PleurX catheter when used to treat recurrent ascites associated with malignancy. J Vasc Interv Radiol. 2008;19(12):1723–31.
- 72. Qu C, Xing M, Ghodadra A, McCluskey KM, Santos E, Kim HS. The impact of tunneled catheters for ascites and peritoneal carcinomatosis on patient rehospitalizations. Cardiovasc Intervent Radiol. 2016;39(5):711–6.
- Tapping CR, Ling L, Razack A. PleurX drain use in the management of malignant ascites: safety, complications, long-term patency and factors predictive of success. Br J Radiol. 2012;85(1013):623–8.
- 74. Valle M, Van der Speeten K, Garofalo A. Laparoscopic hyperthermic intraperitoneal peroperative chemotherapy (HIPEC) in the management of refractory malignant ascites: a multi-institutional retrospective analysis in 52 patients. J Surg Oncol. 2009;100(4):331–4.
- Ba M, Long H, Zhang X, Tang Y, Wu Y, Wang S, et al. Hyperthermic intraperitoneal perfusion chemotherapy and cytoreductive surgery for controlling malignant ascites from ovarian cancer. Int J Gynecol Cancer. 2016;26(9):1571–9.

- 76. Morganti AG, Trodella L, Valentini V, Barbi S, Macchia G, Mantini G, et al. Pain relief with short-term irradiation in locally advanced carcinoma of the pancreas. J Palliat Care. 2003;19(4):258–62.
- Nagels W, Pease N, Bekkering G, Cools F, Dobbels P. Celiac plexus neurolysis for abdominal cancer pain: a systematic review. Pain Med. 2013;14(8):1140–63.
- Lillemoe KD, Cameron JL, Kaufman HS, Yeo CJ, Pitt HA, Sauter PK. Chemical splanchnicectomy in patients with unresectable pancreatic cancer. A prospective randomized trial. Ann Surg. 1993;217(5):447–55; discussion 56–7.
- Allen PJ, Chou J, Janakos M, Strong VE, Coit DG, Brennan MF. Prospective evaluation of laparoscopic celiac plexus block in patients with unresectable pancreatic adenocarcinoma. Ann Surg Oncol. 2011;18(3):636–41.
- Badgwell BD, Cormier JN, Wray CJ, Borthakur G, Qiao W, Rolston KV, et al. Challenges in surgical management of abdominal pain in the neutropenic cancer patient. Ann Surg. 2008;248(1):104–9.
- Glenn J, Funkhouser WK, Schneider PS. Acute illnesses necessitating urgent abdominal surgery in neutropenic cancer patients: description of 14 cases and review of the literature. Surgery. 1989;105(6):778–89.
- Garrett J, Klimberg VS, Anaissie E, Barlogie B, Turnage R, Badgwell BD. The surgical management of abdominal pain in the multiple myeloma patient. Am J Surg. 2012;203(2):127–31.
- Miner TJ, Jaques DP, Tavaf-Motamen H, Shriver CD. Decision making on surgical palliation based on patient outcome data. Am J Surg. 1999;177(2):150–4.
- 84. Badgwell B, Krouse R, Cormier J, Guevara C, Klimberg VS, Ferrell B. Frequent and early death limits quality of life assessment in patients with advanced malignancies evaluated for palliative surgical intervention. Ann Surg Oncol. 2012;19(12):3651–8.
- Badgwell B, Krouse R, Klimberg SV, Bruera E. Outcome measures other than morbidity and mortality for patients with incurable cancer and gastrointestinal obstruction. J Palliat Med. 2014;17(1):18–26.
- Badgwell B, Bruera E, Klimberg SV. Can patient reported outcomes help identify the optimal outcome in palliative surgery? J Surg Oncol. 2014;109(2):145–50.

S. Lamba $(\boxtimes) \cdot R$. Goett

e-mail: lambasa@njms.rutgers.edu

© Springer Nature Switzerland AG 2021

Palliative Care

Sangeeta Lamba and Rebecca Goett

Integrating Emergency Palliative Care for Patients with Advanced Cancer

Seriously ill patients (with malignant and nonmalignant chronic illnesses) who suffer from a high disease and symptom burden often visit the emergency department (ED) for acute crisis events related to their illness [1-13]. Though the ED is primarily designed to resuscitate and stabilize the acutely ill and injured, increasingly those with chronic serious underlying disease processes such as malignancy seek care in this setting [1-5, 7, 10-18]. Malignancy-related symptoms and oncologic emergencies therefore often lead to ED visits, and these visits tend to increase as the patient's clinical status deteriorates and as they approach the end of life [1, 3, 4, 11]. Most hospitalizations in patients with underlying malignancy are initiated from the ED, and these early hours of care often include life-sustaining decisions such as ventilator support and symptom control that establishes, for better or for worse, the trajectory of future in-hospital care [15–20]. When a patient with a life-threatening oncologic emergency presents to the ED, these rapid decisions often occur in the context of uncertain prognosis and rapidly evolving clinical status, especially if the event was unexpected [17-20]. Determining goals of care rapidly so that initial treatment pathways align with patient values (thus avoiding future conflict) is challenging in most circumstances, but perhaps even more so in the ED setting [17-20]. At times, ED providers may need to change gears and shift their focus to comfort and quality of life goals for the patient (palliative care) as opposed to the traditional focus on cure or disease-centered treatment. Strategies to provide optimal care to the seriously ill patient with cancer include (1) use of best practice based clinical decision-making models [21, 22] and (2) incorporation of patients' values and goals in plans of care [15–19].

The World Health Organization defines palliative care as an approach that "improves the quality of life of patients and their families when facing the problems associated with life-threatening illness, through the prevention and relief of suffering by early identification, impeccable assessment and treatment of pain and other problems, physical, psychosocial and spiritual" [23]. Palliative care applies to all phases of a life-limiting condition and is not just for dying patients (Fig. 60.1) [11, 16, 17, 23-25]. In fact, maximal benefit is likely when there is early integration of palliative care into management plans as opposed to only considering such care as a last resort measure when "no more can be done" for the patient [1, 7, 16, 17, 23–31]. The early integration of palliative care is associated with a higher quality of life, including better understanding and communication, access to home care, emotional and spiritual support, wellbeing and dignity, care at time of death, and lighter symptom burden. In fact, some evidence suggests that, on average, palliative care and hospice patients may live longer than similarly ill patients who do not receive such care [6, 17, 32]. Palliative care also has the ability to simultaneously improve quality and control the cost of care for the most seriously ill patients [25, 28, 30-38].

The Institute of Medicine (IOM) report "Dying in America: Improving Quality and Honoring Individual Preferences Near the End of Life" highlights the current disconnect between how most Americans wish to be cared for at the end of their lives and the care that is actually pro-

Simultaneous Care Model

Fig. 60.1 Simultaneous care model of palliative care





Department of Emergency Medicine, Rutgers New Jersey Medical School, Newark, NJ, USA

vided to them [32]. The report lists a similar disconnect between the services that seriously ill patients and families need and the services they currently receive [32]. People nearing the end of life often experience multiple transitions between healthcare settings (the ED being a major setting for end of life crises) and high rates of apparently preventable hospitalizations—which can further fragment the delivery of care and create an added burden for patients and families [32].

The IOM report also makes a recommendation that "All people with advanced serious illness should have access to skilled palliative care or, when appropriate, hospice care in all settings where they receive care (i.e., health care facilities including the ED, the home, and the community)" [32]. This report also proposes that comprehensive care for individuals with advanced serious illness who are nearing the end of life should:

- Be seamless, integrated, patient-centered, family- oriented, and consistently accessible
- Consider the physical, emotional, social, and spiritual needs of individuals and family
- Include coordinated, efficient information transfer across all providers and all settings
- Be consistent with individuals' values, goals, and informed preferences [32].

These proposed goals are consistent with palliative care principles and are integral components of both palliative care and hospice care (Table 60.1) [23, 26, 32]. However, it is important to note that hospice care services in the United States are primarily based on prognosis (as reimbursed by Medicare) and considered when a patient has a terminal prognosis with 6 months or less of predicted survival [16, 17, 20].

Palliative care is "whole person care" or patient-centered care and involves an interdisciplinary team, including board-certified or trained hospice and palliative medicine physicians, nurses, social workers, and chaplains, together with other healthcare professionals as needed [16, 17, 20]. Depending on local resources, access to this team may be on site, via virtual (phone) consultation, or by transfer to a setting with these resources and this expertise level, such as an in-hospital palliative care unit [16, 17, 20]. Although most now recognize the importance of the ED setting in caring for the seriously ill, many barriers and competing priorities for both ED and palliative care providers jeopardize more widespread integration of these two disciplines [16, 17, 39]. The term "integration" is used to indicate the incorporation of palliative care principles (outlined in Table 60.1) [23, 26, 32] into daily ED practice with or without the involvement of a dedicated palliative care team or inpatient palliative care unit.

Table 60.1 Principles and elements of palliative care^a

Patient and family centered care

Care plan is aligned with preferences and determined by goals of patient and family and there is a support system to help family cope with the patient's illness and with bereavement *Timing*

Support starts early in the course of illness and may exist along with therapies that are intended to prolong life (such as chemotherapy) Support continues until disease cure or patient death

Interdisciplinary team approach to care

A team meets the needs of patients and families (may include nurses, social workers, clergy, nursing assistants, pharmacists, and volunteers)

Comprehensive care

Multidimensional assessment to treat physical, psychological, social, and spiritual distress

Relief of suffering

Prevent and relieve suffering from pain and other distressing symptoms

Skills in the care of the dying and bereaved

Prognostication

Offers a support system to help patients live as actively as possible until death

Quality of life

Focus on enhanced quality of life (may also positively influence the course of illness)

Regards dying as a normal process while affirms life

Continuity of care Ensure communication and coordination of care in transitions across settings

Prevent crises and unnecessary transfers are important outcomes Quality assessment and performance improvement^b

Address safety and incorporate the systems of care that reduce error Use validated instruments for data to measure outcomes, when feasible

^aPatient population: patients of all ages experiencing a serious chronic or life-threatening illness or injury

^bCrucial emergency department (ED) and hospital metrics may include ED visits, ED length of stay, time of arrival to time of disposition, hospital and intensive care unit (ICU) length of stay, hospital readmissions, ICU admissions, documenting advance directives, and pain/symptom control Adapted from the National Consensus Project for Quality Palliative Care: Clinical Practice Guidelines for Quality Palliative Care and the World

Health Organization palliative care Definition [23, 26, 32]

Generalist Versus Specialist Emergency Palliative Care [17, 40, 41]

Many palliative care-related skills such as management of pain and other distressing symptoms as well as aligning management with a patient's goals of care are expected to be delivered by all emergency practitioners, so-called generalist practitioner level skills [7, 17, 22, 40, 41]. However, more complex palliative skills such as negotiating a difficult family meeting and managing advanced or refractory symptoms may require a "specialist level" of expertise [7, 17, 22, 40, 41]. Though it may be ideal to have a specialist manage all elements of palliative care, the reality is such that there are
Table 60.2Generalist versus specialist levels of palliative care [40,41]

Generalist-level palliative care Provided by healthcare professionals who manage seriously ill patients, but palliative care is not the main focus of their daily work Includes care in settings not specialized in palliative care such as the emergency department Specialist-level palliative care Provided by healthcare professionals where main activity is the provision of palliative care Includes care in settings specialized in palliative care such as inpatient palliative care unit or hospice

Examples of tasks

Basic management of pain and other distressing symptoms Basic discussions about prognosis, goals of treatment, and advance directives

Management of refractory pain or other difficult-to-treat complex symptoms Conflict resolution regarding goals of care (between family members and between family and healthcare team) Futility of care conversations

not enough specialists in the workforce, to do so. As people live longer with a higher burden of chronic illness, the demand for both generalist and specialist palliative care will rise [40]. Palliative care training programs have expanded nationwide, but the current levels of new trainees in palliative medicine will not meet the needs for all patients who may benefit from such "specialist" care [40, 42]. An optimal care model includes both generalist emergency palliative care (skills that all emergency clinicians should have) and specialist emergency palliative care (skills for managing more complex cases) (Table 60.2) [17, 40, 41]. The Education on Palliative and End-of-life Care (EPEC), EPEC-EM, and End-of-Life Nursing Education Consortium courses are examples of training that seek to build the generalist level of skills for emergency practitioners [43, 44]. Workshops and guides available from multiple resources also target further specific palliative skill development for interested practitioners [45, 46]. Due to the palliative care workforce gap, there is a growing need to develop generalist palliative care skills during emergency medicine resident training and also to provide continuing medical education to practicing clinicians [17, 40]. In addition, some state licensing boards require completion of pain and palliative care education credits prior to license renewal [47]. Performance and quality measure metrics can be used to reinforce the ongoing emphasis on generalist emergency palliative care as an integral component of high-quality patient care [27, 48].

Integrated Emergency Medicine-Palliative Care Initiatives

The Improving Palliative Care in EM (IPAL-EM) project is a resource development and dissemination initiative begun in 2010 by the Center to Advance Palliative Care (CAPC) with funding provided by the Olive Branch Foundation [7, 49,

50]. The goal of this initiative was to accelerate the integration of palliative care services into ED settings. It brought together an advisory panel of nationally recognized leaders in the disciplines of emergency medicine and palliative care [7, 22, 49, 50]. IPAL-EM offers an online portal for sharing essential expertise and available best evidence, tools, and practical resources to assist emergency clinicians and ED administrators in the successful integration of palliative care and EM [22, 49]. Currently, an institutional subscription to CAPC provides access to all the resources gathered as part of the IPAL-EM project, including clinical practice guidelines, needs assessment tools, and ED-specific quality metrics with a relevant library of peer-reviewed consensus/policy statements [49]. Finally, the American College of Emergency Physicians (ACEP) Palliative Care Section provides a variety of online resources to support ED-based palliative care [51].

Demonstration Models of Integrated Emergency Palliative Care

In the last decade, several programs have established ED palliative care initiatives to identify patients that may benefit from early palliative care interventions [22]. The IPAL-EM initiative collected information on existing models or demonstrations of ED and palliative care service integration. Eleven US hospital-based clinical integration programs were interviewed after they were identified from a review of literature, national presentations, and feedback from peer emergency palliative care experts [22]. These programs had varying levels of collaboration between their institutional palliative care program and the ED. Four themes emerged regarding ED palliative care programmatic development including (1) traditional consultation, (2) basic integration, (3) advanced integration, and (4) ED-focused advanced integration models (Fig. 60.2) [22].

Traditional Consultation Programs Similar to other consultation services that interact with the ED, a palliative care expert or consultant is contacted by the ED provider to help answer questions or issues in patient care and to help manage difficult-to-control symptoms [24, 40]. In this model there are no common programmatic goals or process steps to improve overall care delivery [22].

Basic Integration Programs In these models there exists a somewhat more formal relationship between the palliative care program and the ED, and they may work together to achieve some common programmatic goals and objectives. For example, there may be defined protocols for improved patient workflow such as expedited admission to a palliative care unit or targeted generalist-level palliative care education for the ED staff [22].



Intensity of ED Engagement in Palliative Care

Fig. 60.2 Observed models of ED and palliative care integration [22]

Advanced Integration Programs These programs build on the basic integration models to set up common programmatic processes and protocols with the ED increasingly taking on a more active role, while an enhanced level of ED generalist-level palliative care is provided [22, 52].

ED-Focused Advanced Integration Programs In these programs, the ED is very engaged and focused on palliative care-related processes. In some cases the ED may lead the integration and the palliative care services in the institution. Common themes in these programs include case management for high-risk palliative care patients and existence of dual EM and palliative medicine-certified physicians who are passionate about the integration initiatives. Additionally, these programs often have increased numbers of personnel resources to support palliative care, including ED social workers or bereavement supporters for families. They may also implement reorganization and structural changes to improve patient care at the end of life, such as a designated private room or space for imminently dying patients and their families [22, 53].

Jump-Starting an ED Palliative Care Integration Initiative

Researchers have identified a number of important barriers to emergency medicine and palliative care integration. Surveys of physicians, nurses, and administrators list barriers such as (1) the ED culture of life-prolonging care and resuscitation that places a lower emphasis on nontechnical skills, (2) palliative care staffing and availability for the 24-hour, seven days a week, needs of the ED, (3) logistical issues including lack of access to patient medical records, and (4) medicolegal concerns particularly if life-prolonging therapy is not offered [10, 15, 39, 54]. One manuscript describes the ED as being "caught in the middle" when caring for such patients. It lists the challenging physical environment (privacy, noise, lack of information and delay, and lack of defined pathways), with limited resources (overcrowding, time pressures, competition with other emergencies) and variable roles and expectations of the staff providing care (comfort with dying, views of dying in the ED, expertise and comfort in caring for those with serious and advanced illnesses) [10]. On the other hand, the emergency medicine resident and attending physician surveys report that they believe that palliative care skills are important for EM practice but that they are not yet adequately educated and trained in providing such care [15, 55]. Domains of particular interest identified for emergency physician training include management of patients under hospice care, withdrawal of life-prolonging measures, prognostication, and pain management [15, 17, 55]. Recently, hospice and palliative medicine educational milestones were developed for emergency medicine resident training [56].

It is important to begin an ED-PC initiative with identified ED "champions" who can effectively build upon lessons learned from other prior successes and failures so that the initiative is tailored to fit the unique ED setting [7, 50]. The design of the initiative should recognize the preexisting hospital and community resources, availability and hours of access to palliative care consultation services to the ED, and key institutional deficiencies in ED palliative care [7, 50]. Some common examples of initial targets for integration may include:

- Setting up an ED bereavement program
- Defining screening criteria to identify high-risk patients for early palliative care team interventions
- Educating ED staff on pain and palliative care protocols
- Embedding palliative care staff in ED rounds
- Improving throughput to the inpatient palliative care unit, when available

These initiatives have a higher chance of success if aligned with the key metrics important to the institution, such as ED length of stay, hospital readmission rates, and utilization metrics related to observation and intensive care units [7, 17, 48, 50]. Described below are four steps that may help jump-start such an integration initiative [7, 50].

1. Put Together a Team: Palliative Care "Champions" in the ED

Recruiting work group or team members who are interested and committed to the integration of palliative care in the ED is important. For example, identify those individuals who have previously expressed concern, frustration, or sensitivity to a patient's unmet palliative care needs. The interdisciplinary collaborative nature of palliative care allows for engagement of varied professional disciplines based in the ED such as social workers and case managers, as well as other providers throughout the institution who interact with the ED, such as chaplains. Since the integration initiative has the potential to impact other hospital services and processes, a wider range of administrative and clinical personnel should also be considered for inclusion. Table 60.3 lists some of the
 Table 60.3
 Suggested members for the emergency department (ED)

 palliative care integration initiative [7]

ED medical director
ED physician(s)
ED nurse manager and ED nurse(s)
Director or designee of the palliative care program
Nursing educator
Social workers
Case managers
Chaplain
Representatives of key hospital services (e.g., hospitalists, ICU,
surgery, oncologic) that may be affected by this initiative in the ED
Hospital leadership: administration and finance
Others relevant to the success of a specific part of the initiative (e.g.,
ethics consultant, mental health professional, pharmacist)

members to consider when setting up an initial work group. Though the type of members engaged in such an initiative will likely vary from one ED to another, it is vital to include key ED administrators such as the ED medical director and nurse manager(s) in the work group. They not only know their own ED's needs but can also provide valuable perspectives, taking into account resources (staffing, training needs) that may be critical for designing and implementing feasible integration efforts [7, 50]. The ED administrators may also be best equipped to engage support among both ED colleagues and at a broader institutional level.

2. Explore the Resources Available: Existing Literature and Resources

The literature in emergency palliative care is increasing rapidly. Consensus statements on the role of palliative care and ED such as the policy statements and the Choosing Wisely campaign from the American College of Emergency Physicians and Emergency Nurses Association on roles of the ED and ethical issues at end of life are important to note [57–59]. Specific guidance from the Choosing Wisely campaign states: "Don't delay engaging available palliative and hospice care services in the emergency department for patients likely to benefit" [57]. While there is no identified optimal model of ED integration of palliative care, there are a growing number of examples in the literature of specific strategies and programs that have proven to be successful [21, 22, 30, 31, 60]. Other topics relevant to emergency palliative care include:

- Palliative and end-of-life care in the ED [1, 2, 4, 5, 7, 16–20]
- Communication skills [5, 15]
- Family experience and surrogate decision-making [5]
- Palliative care specialists in the ED [19, 40]
- Ethical issues [5, 15]
- Quality improvement/practice change [48]

- Protocols and screening criteria [30, 31, 60]
- Family presence during resuscitation [17]
- Bereavement care [17, 25]
- Education and training [55, 61]

An open-access educational online resource published by the Palliative Care Network of Wisconsin, Fast Facts and Concepts, is worth noting since it provides concise, practical, peer-reviewed, and evidence-based summaries on key topics important to clinicians and trainees caring for patients facing life-limiting illnesses [62]. Fast Facts are free, easily accessible, and clinically relevant monographs on palliative care topics. They are intended as quick teaching tools for bedside rounds, as well as self-study material for trainees and clinicians who work with patients with life-limiting illnesses. For example, separate Fast Fact monographs review protocols for ventilator withdrawal protocol, guidance for calculating opioid dose conversions, and practical aspects of initiating a hospice referral from the ED [62, 63]. In addition, there are several formal educational opportunities in palliative care targeted to ED clinicians. Some programs address overall palliative care skills for many types of providers, while others target profession-specific skill development, for example, those for social work and chaplains. Sponsoring some or all of the identified "champions" at such a conference may help build an institutional pool of qualified candidates who can not only train others but could then be targeted for future career development. Some opportunities include:

- (a) EPEC-EM (Education in Palliative and End-of-life Care for Emergency Medicine): EPEC-EM is a 2-day conference designed to teach clinical competencies in palliative care to ED healthcare professionals in a train-the-trainer format. The conference covers topics specific to ED practice including rapid palliative assessment, disease trajectories and prognosis, care of the hospice patient, chronic and malignant pain management, family-witnessed resuscitation, communication, and more. There is also a focus on techniques for teaching the curriculum to other emergency practitioners [43].
- (b) ELNEC (End-of-Life Nursing Education Consortium): ELNEC offers a modular train-the-trainer end-of-life training program for nurses [44].
- (c) Communication skills building workshops such as Vital Talk [45, 46] that have now developed ED-specific workshops.
- (d) Hospice and palliative medicine fellowship.

The American Board of Emergency Medicine is one of ten sponsoring boards for the hospice and palliative medicine subspecialty. The ACGME provides a program listing and additional information about individual pro- grams that can be accessed online [64].

3. Identify and Ease Access to Local Hospice and Palliative Care Resources

Identify palliative care resources (both personnel and services) that are available: (1) within the ED, such as case managers; (2) in the institution, such as a chaplain, social worker, or bereavement counselor; and (3) in the community such as collaborative arrangements with hospice agencies [30, 31]. These resources are often available, but remain unfamiliar to the ED staff and even if known are not easily identifiable by ED staff at the time of critical need. Collating information and making it easy to access are therefore valuable. Other steps to ease access to palliative resources may include: (1) Identifying and listing various hospital and community resources (Table 60.4); (2) cataloguing their roles, responsibilities, and contact numbers; (3) posting call schedules for personnel in a visible, high traffic area of the ED for ease of access; and (4) identifying clearly the hours of availability of support personnel and whether they are available in person or by phone. Additionally, consider explaining roles and responsibilities of personnel. For example, social work and case managers may be essential partners when navigating disposition issues and maximizing community resource utilization [7].

Palliative Care Consults If the institution has a specialtylevel palliative care team, it may be important to collaborate with them to create screening criteria that assist ED staff in identifying appropriate reasons for consultation (Table 60.5) [7, 60]. Since many such palliative care consultation teams offer in-person services during regular working weekday hours and phone support during off-hours and weekends, it may be useful to create collaborative guidelines to determine what will constitute a non-urgent versus urgent/emergent level of consultation [7, 40, 59].

Inpatient Palliative Care Units Similarly, if the institution has a specialized inpatient palliative care unit, it may be useful to not only establish formal guidelines and processes for admissions from the ED but also educate ED staff on the scope and capabilities of care in this setting. This collaboration

Table 60.4Potential list of institutional and community resources [7, 50]

Palliative care team call schedules
Palliative care team hours of in-person availability
Outpatient palliative care clinic availability and practice hours
Community hospice: home and residential hospice
Chaplaincy support and availability
Social work support and availability
Bereavement support and availability
Ethics consultant
Child life specialist support and availability (for pediatric patients or
children of adult patients)

 Table 60.5
 Screening criteria for a palliative care assessment at the
 time of admission [60] A potentially life-limiting or life-threatening condition (such as malignancy)^a and Primary criteria: Global indicators that represent the minimum that hospitals should use to screen patients at risk for unmet palliative care needs The "surprise question": Would you be surprised if the patient died during this admission? Frequent admissions (admissions for same condition within several months) Difficult-to-control (moderate-severe) physical or psychological symptoms Complex care requirements (functional dependency; home support for ventilator or tube feedings) Decline in function or overall failure to thrive Secondary criteria: Specific indicators that may suggest a high likelihood of unmet palliative care needs Admission from a long-term care facility Metastatic or locally advanced incurable cancer Chronic home oxygen use Out-of-hospital cardiac arrest Current or past hospice program enrollee Limited social support (family stress, chronic mental illness, etc.) No history of completing an advance care planning discussion/ document ^aLife-limiting or life-threatening condition is defined as any disease

^aLife-limiting or life-threatening condition is defined as any disease known to be life-limiting (e.g., chronic obstructive pulmonary disease, metastatic cancer) or that has a high chance of leading to death (e.g., multi-organ failure, sepsis). Serious medical conditions for which recovery to baseline function is routine (e.g., community-acquired pneumonia in a healthy adult) are not included in definition

with the palliative team and ED may be able to prevent some unwanted ED visits, for example, by referrals to an outpatient palliative care clinic.

Hospice Collaborations In addition, appropriate referrals to hospice from the ED are feasible and may facilitate early dispositions [20, 63]. Since these decisions are based on patient-determined goals of care with engagement of loved ones, they have the potential to increase patient and family satisfaction with ED care [20]. Fostering collaborative relationships with local hospice agencies and engaging them in a timely manner for appropriate patients may help clinicians initiate hospice referrals directly from the ED [20]. These relationships also have the potential to improve dialogue when managing patients under hospice care who arrive to the ED with a crisis event related to control of distressing symptoms [20].

4. Complete a Needs Assessment for ED Palliative Care

An assessment of needs helps identify opportunities for improvement in ED palliative care. This may help with targeting of areas where simple interventions can lead to early
 Table 60.6
 Sample section of the needs assessment tool [26, 65]

Domain 2: Physical aspects of care					
Guideline 2.1: ED clinicians use a multidisciplinary approach to pain					
and symptom	m control				
Indicator		Present	Absent	Comment	
2.1	ED clinicians collaborate				
	with specialists from				
	different disciplines to				
	create a comprehensive				
	pain/ symptom control plan				
	of care				
Guideline 2	.2: ED clinicians assess symp	toms usir	ng valida	ted	
assessment	tools appropriate for patients	across the	e life spa	in	
Indicator		Present	Absent	Comment	
2.2.1	Standardized pain				
	assessment tools are used				
2.2.2	Standardized symptom				
	distress assessment tools				
	are used				
Guideline 2.3: Emergency nurses use nurse-initiated protocols to					
relieve the symptom burden of patients					
Indicator		Present	Absent	Comment	
2.3	The ED uses nurse-				
	initiated analgesic				
	protocols				

success that in turn provides momentum to the integration initiative. The needs assessment can outline barriers to the integration initiative and specific institutional strengths and weaknesses and finally identify adherence gaps between best practice guidelines and local practice. This information can focus initial attention and aid decisions on assigning resources [7].

Core guidelines for effective palliative care are outlined by the National Consensus Project for Quality Palliative Care (NCP) and address eight palliative care domains, including physical aspects of care and social aspects of care [26]. These guidelines represent goals and ideal practices that enable programs to define their own palliative program organization, resource requirements, and performance measures. These guidelines have been adapted to develop ED-specific clinical practice guidelines and translated into a "needs assessment tool" that programs may find useful to identify areas for improvement (Table 60.6) [65].

Monitoring Integrated Palliative Care Initiatives

The Agency for Healthcare Research and Quality (AHRQ)/ American College of Emergency Physicians conference on "Improving the Quality and Efficiency of Emergency Care Across the Continuum: A Systems Approach" identified four key topics or questions for emergency medicine and palliative care: (1) Which patients are in greatest need of palliative care services in the ED (identifying the target population in
 Table 60.7
 Potential measurable quality metrics and outcomes for an emergency department (ED) palliative care integration initiative

Operational

Mean/median ED length of stay (hours)

Discharge disposition status

ED arrival to time of disposition (to palliative care unit)

% With repeat ED visits within 30 days (within 60, 90 days) % With repeat hospital admits within 30 days (within 60, 90 days) Number of hospice referrals from the ED

Number of palliative care referrals from the ED (if available) *Clinical*

% Charts with documentation of the healthcare decision-maker/ advance care directives

% Of patients prescribed opioids with bowel regimen on discharge

% Of patients with documented pain assessment on presentation and % reassessed

% Of families with documented offer of spiritual support after ED death

% Of patients in target populations who have a documented palliative care assessment

% Of caregivers in target patient populations screened for caregiver strain

Patient satisfaction

% Of ED patients who report being adequately informed about their condition or treatment plans and options

% Of families who report excellent overall end-of-life care after patient's ED death

% Of patients reporting satisfaction with communication regarding discharge instructions

need)? (2) What is the optimal role of emergency clinicians in caring for patients along a chronic trajectory of illness (what skills are necessary)? (3) What is the effect on healthcare utilization after the integration and initiation of palliative care training and services in the ED? and (4) What are the educational priorities for emergency clinicians in the domain of palliative care? The conference proposed that future emergency palliative care research gathers evidence in these domains using six categories of inquiry: descriptive; attitudinal; screening; outcomes; resource allocation; and education of clinicians [48, 66]. Examples of some relevant quality indicators to measure progress toward ED-palliative care integration are listed in Table 60.7. Emerging work is exploring the efficacy of decision support tools and accuracy of screening triggers and tracking outcomes of the initiatives described above [67, 68].

Health Economics

Building palliative care capacity or launching new integrated palliative care initiatives in the ED may have costs that can range from minimal to substantial. For example, if the goal is to identify patients with palliative care needs early in the ED, an approach that uses preexisting resources such as the ED triage nurse to screen will have nominal costs, whereas if a dedicated staff such as a social worker will perform initial screening, the cost will be substantial. Therefore, leveraging preexisting resources such as nurse and faculty educators for educational initiatives and sharing interdisciplinary team members such as social work across units may allow for significant success without a large financial impact. Use of qualitative assessment (QA) EMR consult triggers, documentation templates, or order sets are also relatively inexpensive to place but need promotion and education around implementation.

For major and advanced palliative care integration initiatives, the main cost is personnel related, and buy-in from the administration would be needed. Highlighting elements that are important to the institution such as the fact that palliative care has been shown to increase patient satisfaction, improve quality of life, and decrease number of ICU days (and therefore related costs) may be useful to make the case and the value proposition for funding palliative care [6, 17, 25, 28, 30-38]. Leveraging philanthropy may offer ways to get smaller seed funds to jump-start an initiative as well. Education of clinicians around existing re-imbursement pathways may also incentivize good practices. For example, in 2018, Centers for Medicare & Medicaid Services updated their policy for providers to code for reimbursement when they provide advanced care planning and goals of care discussions with patients.

Case Studies

Case 1 Communication Skills: Delivery of Difficult or Serious News

A 67-year-old woman with hypertension and a history of ovarian cancer with surgical resection 7 years ago presents to the ED with moderate shortness of breath. Since the patient is afebrile and has unexplained tachycardia, she has a CT scan of the chest performed to rule out a pulmonary embolism. Multiple large pulmonary nodules suspicious of metastatic disease are seen. This disease progression is new based on patient's records.

ED Tasks Communication of serious or difficult news, advance care planning, aligning goals of care, symptom control, psycho-social and emotional support, and disposition to hospital or palliative care unit.

Case 2: Rapid Goals of Care Discussion

A 68-year-old woman with diabetes and breast cancer with metastatic bone disease is undergoing palliative radiation for pain management. She has severe fatigue, fever, and shortness of breath with oxygen saturation of 87% on 15-liter non-rebreather mask and high flow nasal cannula. Her chest X-ray shows diffuse bilateral ground glass infiltrates consistent with coronavirus infectious disease 2019. She is awake, anxious, and struggling to speak more than two to three words. She wants her live-in boyfriend of 10 years to make decisions regarding intubation but does not want her only daughter engaged. She also says that if she stops breathing, "that's God's will."

ED Task Effective communication of serious news, rapid goals of care, identifying healthcare proxy, managing possible family conflict, ethical issues, symptom management.

Case 3: Symptom Management: Actively Dying Patient

Patient is an 83-year-old man with congestive heart failure NYHA Class 4 and prostate cancer with bony metastases and presents to the ED for acute respiratory failure and hypotension with septic shock. He is enrolled in hospice at home, but his wife got anxious because he was struggling to breathe and she says, "please help him." Patient has a do-notresuscitate and do-not intubate order on file.

ED Tasks Symptom management, aligning goals of care, coordinating disposition, transitioning care to in-patient hospice and/or palliative care unit, bereavement, and spiritual and psychosocial support of caregiver.

Summary

Optimal care for the seriously ill patient with cancer in the emergency department includes an early integration of palliative care for eligible patients. This capacity for palliative care is built by fostering the skills and competencies for the generalist ED clinician who will provide much of this care to meet patient needs such as effective communication of serious news, aligning treatment with patient's goals of care and wishes, and managing distressing symptoms. Capacity is further built using an integrated approach with early identification of patients with complex palliative care needs who would benefit from specialist-level consultation in the ED and navigating patient and family to appropriate palliative care and hospice resources. Integration initiatives in the ED may have a higher chance of success when ED champions are fully engaged with palliative care experts to collaboratively define not only resources and processes but also appropriate metrics to track outcomes and measure impact of the integrated initiative.

References

- Barbera L, Taylor C, Dudgeon D. Why do patients with cancer visit the emergency department near the end of life? CMAJ. 2010;182(6):563–8.
- Vandyk AD, Harrison MB, Macartney G, Ross-White A, Stacey D. Emergency department visits for symptoms experienced by oncology patients: a systematic review. Support Care Cancer. 2012;20(8):1589–99.
- Earle CC, Neville BA, Landrum MB, Ayanian JZ, Block SD, Weeks JC. Trends in the aggressiveness of cancer care near the end of life. J Clin Oncol. 2014;22(2):315–21.
- Leak A, Mayer DK, Wyss A, Travers D, Waller A. Why do cancer patients die in the emergency department? An analysis of 283 deaths in NC EDs. Am J Hosp Palliat Med. 2013;30(2):178–82.
- Smith AK, Schonberg MA, Fisher J, Pallin DJ, Block SD, Forrow L, et al. Emergency department experiences of acutely symptomatic patients with terminal illness and their family caregivers. J Pain Symptom Manag. 2010;39(6):972–81.
- Temel JS, Greer JA, Muzikansky A, Gallagher ER, Admane S, Jackson VA, et al. Early palliative care for patients with metastatic non-small-cell lung cancer. N Engl J Med. 2010;363(8):733–42.
- Lamba S, DeSandre PL, Todd KH, Bryant EN, Chan GK, Grudzen CR, et al. Integration of palliative care into emergency department: the IPAL EM collaboration. J Emerg Med. 2014;46(2):264–70.
- Hui D, Kim YJ, Park JC, Zhang Y, Strasser F, Cherney N, et al. Integration of oncology and palliative care: a systematic review. Oncologist. 2015;20:77–83.
- Nauck F, Alt-Epping B. Crises in palliative care—a comprehensive approach. Lancet Oncol. 2008;9(11):1086–91.
- Jelinek GA, Marck CH, Weiland TJ, Philip J, Boughey M, Weil J, et al. Caught in the middle: tensions around the emergency department care of people with advanced cancer. Emerg Med Australas. 2013;25(2):154–60.
- Burge F, Lawson B, Johnston G. Family physician continuity of care and emergency department use in end-of-life cancer care. Med Care. 2003;41(8):992–1001.
- 12. Wiese CH, Vossen-Wellmann A, Morgenthal HC, Popov AF, Graf BM, Hanekop GG. Emergency calls and need for emergency care in patients looked after by a palliative care team: retrospective interview study with bereaved relatives. BMC Palliat Care. 2008;7(1):11.
- Kenen J. Palliative care in the emergency department: new specialty weaving into acute care fabric. Ann Emerg Med. 2010;56(6):A17–9.
- Chan GK. End-of-life models and emergency department care. Acad Emerg Med. 2004;11:79–86.
- Smith AK, Fisher J, Schonberg MA, Pallin DJ, Block SD, Forrow L, et al. Am I doing the right thing? Provider perspectives on improving palliative care in the emergency department. Ann Emerg Med. 2009;54(1):86–93, 93.e1.
- Lamba S, Mosenthal AC. Hospice and palliative medicine: a novel subspecialty of emergency medicine. J Emerg Med. 2012;43(5):849–53.
- Quest TE, Marco CA, Derse AR. Hospice and palliative medicine: new subspecialty, new opportunities. Ann Emerg Med. 2009;54(1):94–102.
- Lamba S. Early goal-directed palliative therapy in the emergency department: a step to move palliative care upstream. J Palliat Med. 2009;12:767.
- Meier DE, Beresford L. Fast response is key to partnering with the emergency department. J Palliat Med. 2007;10:641–5.
- Lamba S, Quest TE, Weissman DE. Initiating a hospice referral from the emergency department #247. Fast Facts and Concepts. J Palliat Med. 2011;14(12):1346–7.

- Mahony SO, Blank A, Simpson J, Persaud J, Huvane B, McAllen S, et al. Preliminary report of a palliative care and case management project in an emergency department for chronically ill elderly patients. J Urban Health. 2008;85(3):443–51.
- 22. Quest TE, Herr S, Lamba S, Weissman DE. Demonstrations of clinical initiatives to improve palliative care in the emergency department: a report from the IPAL-EM initiative. Ann Emerg Med. 2013;61(6):661–7.
- World Health Organization. Definition of palliative care. World health organization web site. https://www.who.int/news-room/factsheets/detail/palliative-care. Accessed 1 Jul 2020.
- Morrison RS, Meier DE. Clinical practice. Palliative care. N Engl J Med. 2004;350(25):2582–90.
- Gelfman LP, Meier DE, Morrison RS. Does palliative care improve quality? A survey of bereaved family members. J Pain Symptom Manage. 2008;36(1):22–8.
- 26. National Consensus Project for Quality Palliative Care. Clinical practice guidelines for quality palliative care. 2nd ed. Pittsburgh: National Consensus Project; 2009. https://www.nationalcoalitionhpc.org/wp-content/uploads/2017/04/NCP_Clinical_Practice_ Guidelines_3rd_Edition.pdf. Accessed 1 Jul 2020.
- Twaddle ML, Maxwell TL, Cassel JB, Liao S, Coyne PJ, Usher BM, et al. Palliative care bench- marks from academic medical centers. J Palliat Med. 2007;10(1):86–98.
- Meier DE. Increased access to palliative care and hospice services: opportunities to improve value in health care. Milbank Q. 2011;89(3):343–80.
- 29. Fadul N, Elsayem A, Palmer JL, Del Fabbro E, Swint K, Li Z, et al. Supportive versus palliative care: what's in a name? A survey of medical oncologists and mid-level providers at a comprehensive cancer center. Cancer. 2000;115(9):2013–21.
- Glajchen M, Lawson R, Homel P, DeSandre P, Todd KH. A rapid two-stage screening protocol for palliative care in the emergency department: a quality improvement initiative. J Pain Symptom Manag. 2011;42(5):657–62.
- Lawson BJ, Burge FI, McIntyre P, Field S, Maxwell D. Can the introduction of an integrated service model to an existing comprehensive palliative care service impact emergency department visits among enrolled patients? J Palliat Med. 2009;12(3):245–52.
- 32. Institute of Medicine. Dying in America: improving quality and honoring individual preferences near the end of life. Washington, DC: The National Academies Press; 2015. https://doi. org/10.17226/18748.
- Morrison RS, Penrod JD, Cassel JB, Caust-Ellenbogen M, Litke A, Spragens L, et al. Cost savings associated with US hospital palliative care consultation programs. Arch Intern Med. 2008;168(16):1783–90.
- Morrison RS, Dietrich J, Ladwig S, Quill T, Sacco J, Tangeman J, et al. Palliative care consultation teams cut hospital costs for Medicaid beneficiaries. Health Aff. 2011;30(3):454–63.
- Penrod JD, Deb P, Luhrs C, Dellenbaugh C, Zhu CW, Hochman T, et al. Cost and utilization outcomes of patients receiving hospitalbased palliative care consultation. J Palliat Med. 2006;9(4):855–60.
- 36. Brumley R, Enguidanos S, Jamison P, Seitz R, Morgenstern N, Saito S, et al. Increased satisfaction with care and lower costs: results of a randomized trial of in-home palliative care. J Am Geriatr Soc. 2007;55(7):993–1000.
- 37. Bakitas M, Lyons KD, Hegel MT, Balan S, Brokaw FC, Seville J, et al. Effects of a palliative care intervention on clinical outcomes in patients with advanced cancer: the Project ENABLE II randomized controlled trial. JAMA. 2009;302(7):741–9.
- 38. Connors AF, Dawson NV, Desbiens NA, Fulkerson WJ, Goldman L, Knaus WA, et al. A controlled trial to improve care for seriously ill hospitalized patients: the study to understand prognoses and preferences for outcomes and risks of treatments (SUPPORT). JAMA. 1995;274(20):1591–8.

- Lamba S, Nagurka R, Zeilinski A, Scott SR. Palliative care provision in the emergency department: barriers reported by emergency physicians. J Palliat Med. 2013;16(2):143–7.
- Quill TE, Abernethy AP. Generalist plus specialist palliative care— creating a more sustainable model. N Engl J Med. 2013;368(13):1173–5.
- Gamondi C, Larkin P, Payne S. Core competencies in palliative care: an EAPC white paper on palliative care education: part 2. Eur J Palliat Care. 2013;20(3):140–5.
- Morrison RS, Maroney-Galin C, Kralovec PD, Meier DE. The growth of palliative care programs in United States hospitals. J Palliat Med. 2005;8(6):1127–34.
- Emanuel LL, Ferris FD, Von Gunten CF. EPEC. Education for physicians on End-of-Life Care. Am J Hosp Palliat Care. 2002;19:17.
- 44. End-of-Life Nursing Education Consortium. http://www.aacn. nche.edu/elnec. Accessed 2 Jul 2020.
- Vital talk. Quick guides. http://www.vitaltalk.org/quick-guide. Accessed 2 Jul 2020.
- Onco Talk Learning Modules. https://depts.washington.edu/toolbox/content.pdf. Accessed 2 Jul 2020.
- Walther Center in Global Palliative Care & Supportive Oncology. U.S. resources. Federation of state medical boards. https://walthercenter.iu.edu/resources/us-resources/index.html. Accessed 9 Oct 2020.
- 48. Agency for Healthcare Research and Quality. HRQ quality indicators. https://www.qualityindicators.ahrq.gov/Downloads/ Resources/Publications/2006/EDPerformanceMeasures-ConsensusStatement.pdf. Accessed 2 Jul 2020.
- Improving Palliative Care in Emergency Medicine; IPAL-EM project. Center to advance palliative care. https://www.capc.org/ipal/ ipal-emergency-medicine/. Accessed 2 Jul 2020.
- 50. Lamba S, Schmidt T, Grudzen C, Chan GK, Todd K, Quest TE. Integrating palliative principles in the out-of-hospital setting: four things to jump start an EMS-palliative care initiative. Prehosp Emerg Care. 2013;17(4):511–20.
- American College of Emergency Physicians. "ACEP: palliative care section" ACEP.org. https://www.acep.org/how-we-serve/sections/palliative-medicine/. Accessed 2 Jul 2020.
- Fine RL. The imperative for hospital-based palliative care: patient, institutional, and societal benefits. Proc (Bayl Univ Med Cent). 2004;17(3):259–64.
- Rosenberg M, Rosenberg L. Integrated model of palliative care in the emergency department. West J Emerg Med. 2013;14(6):633–6.
- 54. Grudzen CR, Richardson LD, Major-Monfried H, Kandarian B, Ortiz JM, Morrison RS. Hospital administrators' views on barriers and opportunities to delivering palliative care in the emergency department. Ann Emerg Med. 2013;61(6):654–60.
- Lamba S, Pound A, Rella JG, Compton S. Emergency medicine resident education in palliative care: a needs assessment. J Palliat Med. 2012;5(5):516–20.
- 56. Shoenberger J, Lamba S, Goett R, DeSandre P, Aberger K, Bigelow S, et al. Development of hospice and palliative medicine knowledge and skills for emergency medicine residents: using the accreditation council for graduate medical education milestone framework. AEM Educ Train. 2018;2(2):130–45. https://doi.org/10.1002/aet2.10088.
- 57. Choosing Wisely Campaign. http://www.choosing-wisely. org/nations-emergency-physicians-announce-list-of-test-andprocedures-to-question-as-part-of-choosing-wisely-campaign/. Accessed 2 Jul 2020.
- American College of Emergency Physicians. Ethical issues in emergency department care at the end of life. Ann Emerg Med. 2006;47(3):302.
- Emergency Nurses Association (ENA). Position statement on endof-life care in the emergency department. https://www.ena.org/ docs/default-source/resource-library/practice-resources/position-

statements/palliativeendoflifecare.pdf?sfvrsn=1777bb45_8. Accessed 2 Jul 2020.

- Weissman DE, Meier DE. Identifying patients in need of a palliative care assessment in the hospital setting: a consensus report from the Center to Advance Palliative Care. J Palliat Med. 2011;14(1):17–23.
- Gisondi MA. A case for education in palliative and end-of-life care in emergency medicine. Acad Emerg Med. 2009;16:181–3.
- Palliative Care Network of Wisconsin. Fast facts and concepts. Marks S, editor. https://www.mypcnow.org/fast-facts/ Accessed 9 Oct 2020.
- Lamba S, Quest TE, Weissman DE. Initiating a hospice referral from the emergency department #247. Fast facts and concepts. J Palliat Med. 2011;14(12):1346–7.
- Accreditation Council for Graduate Medical Education (ACGME)—Public. Program search, hospice and palliative medicine fellowship programs. https://apps.acgme.org/ads/Public/ Programs/Search. Accessed 2 Jul 2020.

- 65. Chan GK, Bryant EN, Lamba S, Quest TE, Weissman DE, Todd KH. Clinical practice guidelines-self assessment tool. A technical assistance resource from the IPAL-EM project. Center to advance palliative care. https://www.capc.org/ipal/ipal-emergencymedicine/. Accessed 2 Jul 2020.
- 66. Quest TE, Asplin BR, Cairns CB, Hwang U, Pines J. Research priorities for palliative and end-of-life care in the emergency setting. Acad Emerg Med. 2011;18(6):e70–6.
- 67. Tan A, Durbin M, Chung FR, Rubin AL, Cuthel AM, McQuilkin JA, Group authorship PRIM-ER Clinical Informatics Group, et al. Design and implementation of a clinical decision support tool for primary palliative care for emergency medicine (PRIM-ER). BMC Med Inform Decis Mak. 2020;20(1):13.
- 68. Ouchi K, Strout T, Haydar S, Baker O, Wang W, Bernacki R, et al. Association of emergency clinicians' assessment of mortality risk with actual 1-month mortality among older adults admitted to the hospital. JAMA Netw Open. 2019;2(9):e1911139.

Hospice

Bonnie K. Marr and Danielle J. Doberman

Why This Chapter Is Important

Most emergency physicians can recall a patient presenting to the emergency department (ED) with severe pain, dyspnea, or other symptoms and having to give the gut-wrenching news of a new cancer diagnosis or significant disease progression [1, 2]. With sensitive discussion, the patient may reveal they were referred to hospice services by their outpatient specialist for management of their symptom burden or extensive disease progression and ultimately declined to follow these recommendations. Even more memorable is the patient who is currently receiving hospice services yet presents to the ED actively dying. These scenarios are common and stand out due to the complexity of untangling the patient and family's true goals when presenting to the ED. As a clinician new to a patient's journey with serious illness, negotiating the intricacies of "breaking bad news" or navigating goals of care—especially for a patient who may be actively dying-in a time-constrained environment with limited information available is unlike other areas of medicine. However, emergency providers are in a unique and important position to make an impact on the patient's overall illness trajectory and care plan. This chapter will provide additional tools for navigating these circumstances and highlight the progress that can be made despite the challenges of the ED.

Cancer continues to be the leading diagnosis for those enrolled in hospice according to the National Hospice and Palliative Care Organization, with 30.1% of hospice decedents having a principal diagnosis of cancer in 2017 [3]. Therefore, within this text, it is important to emphasize that emergency physicians should be facile in communicating the

B. K. Marr (🖂)

D. J. Doberman

differences between hospice and non-hospice palliative care (PC), evaluating the reasons a patient may present to the ED while enrolled in hospice services, confirming patient and family goals of care, and utilizing available resources to insure a smooth transition across settings.

Background

The Center to Advance Palliative Care (CAPC) defines palliative care as "... specialized medical care for people with serious illnesses... Palliative care is appropriate at any age and at any stage in a serious illness and can be provided together with curative treatment" [4]. Serious illness may include diagnoses, such as cancer, cardiac disease, respiratory disease, kidney failure, dementia, HIV/AIDS, amyotrophic lateral sclerosis (ALS), and many other serious and progressive illnesses. CAPC emphasizes that palliative care should not be initiated based on a patient's prognosis alone, but rather on the need for specialized, coordinated medical care for patients living with any stage of a serious illness from diagnosis onward, and can be provided along with curative treatments. CAPC further defines the overall goals of palliative care to be improved quality of life for the patient and family through assisting with management of symptoms and enhanced coping, which often can include clarification of goals of care [4, 5].

Emergency physicians who engage in reviewing goals of care with a patient and/or family can find it challenging to describe hospice and compare and contrast its services to those offered by palliative care. Both fields share a common philosophy of care that prioritizes patient-centered goals, focuses on the patient and family adaptation to illness and coping, and offers proactive symptom management. Hospice, however, is a program of services for patients who have a prognosis of 6 months or less and is provided to those who forgo further curative or life-prolonging care to focus on dignity and comfort at the end of life [6].



Section of Palliative Medicine, Johns Hopkins Hospital, Baltimore, MD, USA e-mail: bmarr1@jhmi.edu

Section of Palliative Medicine, Department of Emergency Medicine, Johns Hopkins Hospital, Johns Hopkins University, Baltimore, MD, USA

Under hospice care, a physician-signed "Certification of Terminal Illness" (CTI) is required, whereby that physician attests that the patient's prognosis is anticipated to be less than 6 months if the terminal illness were to continue on the anticipated trajectory. Emergency physicians may be asked to sign this form when making referrals to hospice from the ED. At the time of hospice enrollment, the patient designates their Medicare or other insurance benefits to hospice care rather than ongoing curative treatment for the terminal illness and/or related conditions [7]. A helpful resource to review when exploring hospice designation for providers and patients can be found at https://www.medicare.gov/coverage/hospice-care. In addition, the assessment may be complex in supporting the CTI, and your ED case manager or hospice medical director may have special expertise in reviewing this documentation.

When a Patient Enrolled in Hospice Presents to the Emergency Department

Patients presenting from a hospice model of care to the ED are often experiencing a crisis, such as uncontrolled or poorly controlled symptoms, new symptoms such as a seizure or bleeding, or caregiver exhaustion due to the demands of care for someone with a progressive disease [1, 8, 9]. Symptom control needs may simply outpace the capacity of the home environment. On arrival to the ED, patients and their family and support systems do not necessarily wish to revoke hospice. Rather, they seek the rapid assessment and provision of symptom control EDs are known to routinely offer. In the setting of duress, a loved one may call for emergency medical services because this is a familiar way to seek urgent medical care. This type of presentation represents a critical juncture to confirm goals of care and an opportunity to coordinate treatment with the hospice of the patient's choice and align care with the patient's wishes. Counseling the family regarding the patient's current clinical status or recent decline and evaluation for potential refractory symptoms is often required. In addition, as outlined in Table 61.1, coordination of care with the hospice of their choice to ensure needs are being met, expectations are managed, and clear communication provided may prevent hospital admission or a return to the ED.

Given the variety of life-prolonging treatments available in the ED setting, and procedures such as thoracentesis or addressing an unstable fracture, which can dramatically improve comfort for patients even in their last days, yet can carry iatrogenic risks, conversations to urgently confirm and clarify goals of care and discuss alternatives provided by the hospice should take place whenever possible before a plan of care is finalized. Patient-defined goals of care should dictate any testing and treatment and take precedence over ED
 Table 61.1
 Reasons hospice patients present to the emergency department [1, 9]

Patient and support system	Uconico coro acordination reasons
Symptom control not optimized and felt to be intolerable (e.g., pain regimen felt to be ineffective despite titration by hospice)	Inability to address patient needs in a timely manner to improve comfort/lessen suffering
Urgent new symptom (e.g., seizure, hemoptysis, urinary retention, impaction, epistaxis, fall with fracture, or need for suture/cast)	Poor communication and/or anticipatory guidance and education with the patient and family
Medical equipment or supportive device malfunction (e.g., gastrostomy tube dislodgement, oxygen supply running out)	Hospice arranged for ED presentation to assist with supportive device/medical equipment restoration in keeping with patient's advance directives
Patient and caregiver(s) have differing/conflicting philosophies of care	Equipment malfunction/failure without timely repair or replacement to maintain patient's comfort (e.g., nebulizer, suction device, home oxygen, BiPAP)
Caregiver fatigue or challenge meeting the demands of maintaining the patient's comfort/needs	Family new to hospice/hospice experience and unaware to telephone hospice for their needs
Conflict about life-prolonging treatments. These may be treatments that were discontinued (e.g., chemotherapy) or interventions that were never initiated (e.g., hemodialysis)	Patient traveling between hospice catchment areas with unanticipated symptoms and not sure how to obtain hospice assistance
Automatic response to patient distress, a loved one may call EMS because this is a familiar way to seek urgent medical care Patient has enrolled in hospice requesting full resuscitation and EMS activated in keeping	Presenting from a skilled nursing facility where staff is to be directed by hospice team to provide care and staff unfamiliar with hospice procedures/policies without proper training and require urgent assistance

protocols as these may no longer be appropriate in the context of the patient's wishes. The hospice where the patient is enrolled likely has information regarding advance directives, previous intended plan of care, and prior wishes of the patient. They may even be able to send personnel to the ED who can be helpful in further exploration of the patient's goals as well.

Time for the Talk: Introducing Hospice

As outlined earlier in this chapter, patients presenting to the ED with an oncologic diagnosis may qualify for hospice services, including with discovery of disease progression or functional decline; however, clinicians may be hesitant to broach the topic given the constraints of a time-limited interaction and perceived difficulty in coordinating this transition in care [10]. Emergency providers should feel empowered to assess a patient's hospice eligibility and review this with the patient's primary physician, often an oncologist, cardiologist, or other specialist. In introducing hospice, it is necessary to review a patient's understanding of their clinical situation, sometimes called their "prognostic understanding" [11]. Having a shared understanding of prognosis between all providers, patient, and family is key to a successful hospice referral. Next, a provider would review goals of care and ascertain if the patient's wishes are aligned with hospice philosophy of care. Many of the steps in determining the appropriate level of hospice care and reviewing what services can be provided through the hospice are nuanced and based on the patient's clinical status and wishes. Table 61.2 provides a quick reference for overview of the hospice referral process, and Table 61.3 reviews the levels of hospice care.

It is often helpful to include ED social work and case management as early as possible in the care of a patient for whom hospice is being entertained to help in exploring possible options. It may also be useful to discuss the patient directly with the medical director of the hospice of the patient's/family's choice for further coordination of care. While it may seem there are a lot of moving parts to coordinate transition to hospice from an ED visit, this may be an ideal opportunity to facilitate the patient's goals and help the patient and family negotiate a crossroads in their healthcare journey. When there is a pre-existing relationship with the local hospices within the hospital's catchment area and the ED and hospital, the process of direct admission into hospice care from the ED can work smoothly. In these circumstances, social work and case management teams likely have experience offering the patient and family choices of hospice vendors with varying capacities and then facilitating assessment by the hospice(s) of their choice. The ideal scenario is a transfer to either a hospice inpatient unit (IPU) directly from the ED or the arrangement of a hospice intake visit (initial visit by a nurse at the patient's home or place of residence who will admit the patient to hospice and review policies/ procedures and medications for symptom control) on the same day as discharge from the ED.

However, despite these efforts, some patients and family may decline to enroll in hospice from the ED, preferring a discharge home without services. Hitting the proverbial pause button on a disease-modifying plan of care that may not be in alignment with the patient's values and wishes for the future they envision can be an important first step in developing a treatment plan which supports their wishes. Due to the time-compressed environment of the ED, connecting with the patient's primary care physician or primary physician of record (cardiologist, oncologist, etc.) to convey the need for further goals of care discussions and plan for further in-depth discussions in the outpatient setting can facilitate a smooth transition of care. The primary physician may wish to hold these discussions themselves or may request an outpatient palliative clinic consultation be arranged from the ED. If the patient/family are strongly considering hospice and have not yet confirmed this choice, another option is to arrange a hospice informational visit, during which a representative from the local hospice of the patient's choice visits the patient at bedside or in their home to explain the program and its services. This is an ideal opportunity to assist the patient and their family in the information-gathering stages.

Review of Emergency Department and Palliative Care Collaborations: Toward Developing Pathways for Efficient Transfer to Hospice Care

For ED patients who are eligible for hospice, collaboration between ED providers and the hospital's palliative care service may be useful to identify patients who would benefit from a palliative care consult, provide over-arching support for seeking out goal-concordant care, and assist with care coordination with local hospices as appropriate. Quest and colleagues developed the Improving Palliative Care in Emergency Medicine (IPAL-EM) curriculum and not only identified an algorithm for considering palliative care involvement but also review a stepwise process to promote emergency medicine and palliative care collaboration, including the use of a needs analysis [12, 13]. Each institution may differ in the results of the analysis, especially as the viewpoints of various stakeholders are considered; however, through CAPC, this tool kit is available to tailor collaboration based on the needs identified. In terms of hospice, there are opportunities to streamline communication between ED staff and local hospices as well as utilize the local strengths of the ED interdisciplinary team, including social work and case management, to make direct hospice referrals. Patients and families must have the freedom of choice regarding which hospice they choose, and staff should be educated to provide these options and assist with coordination of care. An ED with an established PC partnership may use standard mutually agreeable screening tools to identify patients with unmet palliative care needs for PC consultation and hospice eligibility. A number of screening tools exist to identify these patients [14-16].

Early PC involvement in the ED presents the possibility for increased efficiency in the form of decreased length of hospital stay, improved patient/family satisfaction, decreased ICU utilization, and increased direct hospice referrals [16– 20]. Further, PC programs often have relationships with hospices within a hospital's catchment area and can build

		-	
Steps to consider	Why this is important	Potential resources to assist	Nota bene
Determine hospice eligibility	Patients must qualify for hospice based on general Medicare guidelines used in conjunction with disease-specific considerations	Remember "Certification of Terminal Illness" is required where the patient's prognosis is anticipated to be less than 6 months if the terminal illness were to continue on the anticipated trajectory with specific criteria required [6]. The criteria can be complex and generally provide evidence for the clinical judgment above with factors related to demonstrating overall decline in clinical status versus non-disease-specific and disease-specific guidelines. It is recommended to review if a patient is eligible for hospice with their provider of record and the hospice of choice [11]	Hospice and palliative care teams are great resources for information
Discuss with provider of record	Can confirm hospice eligibility, review remaining treatment options and current treatment regimen, coordinate care, and engage provider with trusted rapport with the patient	Oncologist or primary care doctor may be the point of contact. They also may wish to be the provider of record for the hospice	This provider may also want to speak to the patient/family directly, and therefore it is important to include this person in the beginning for care coordination
Elicit patient information preferences, understanding, and goals of care	Prior to introducing hospice, it is important to find out what information the patient and/ or family wants to receive (e.g., some may not want to know specific prognosis), assess understanding of their status to better manage expectations and elucidate gaps in communication/knowledge, and <i>further</i> <i>establish specific patient-centered goals of</i> <i>care</i>	Please see earlier chapter on communication for specific strategies and resources	Consider palliative care consult
Review philosophy of hospice if compatible with goals of care	Review that hospice is not a specific "place," but a philosophy of care focused on comfort and dignity at the EOL. May be helpful to ask the patient's understanding of hospice or prior experiences to help directly address concerns	Helpful to have social work and case management involved in discussions and to assist in reviewing available care in the home if this is the patient's preference and clinically appropriate	Work with your department to have a ready list of local hospices for ease of providing written materials to patients/families
Hospice choices/ facilitate coordination of care	SW/CM may be helpful in reviewing local hospice resources to provide patients and families with choices available. Once this is done, hospice liaisons may be involved to evaluate the case	It can be helpful to review the case with the medical director of the hospice of the patient's choice as hospices have varying policies/procedures regarding certain interventions, such as IVF, TPN, antibiotics, etc. This is also a chance to review level of care needs—patients must meet certain criteria to be admitted to inpatient units (IPUs)	May be helpful to have list of contact information for liaisons/medical directors for local hospices available for providers
Disposition preparation	What you will need to do: Generally, hospices will require a "Certification of Terminal Illness" and decisions regarding hospice provider of record (of note, this can be the hospice medical director). A finalized MOLST/POLST dependent on the state laws may be needed	What needs to be arranged with the hospice: Arrangement with the hospice for delivery of DME, hospice intake visit, and review of prescriptions that may be needed until a patient's enrollment in hospice is complete	Care coordination with the patient's primary provider, especially if transition to hospice is not possible at time of determination of disposition
Clear sign out!	All the time invested in discussions with the patient, family, hospice, and primary provider should be thoroughly discussed at sign out and carefully documented. Discussions and family meetings are important and like a procedure should be recorded	Given the intense nature of these discussions and the shift nature of most ED provider's schedules, consider introducing the new provider coming on shift to the patient and their family, if possible, to establish continuity and provide clear communication	As in any other discharge, a patient should be assessed for stability for transport and anticipatory guidance given to families as appropriate if there is concern for potential death during transport

 Table 61.2
 Coordinating transition to hospice from the emergency department [1]

Level of care	Description	Considerations	Helpful facts
Routine	<i>Routine care</i> is hospice care that patients have caregivers who can provide around the clock care if needed, have stable symptoms or symptoms anticipated to be controlled utilizing measures accessible <i>within the home, or home-like settings</i> , such as an assisted living, group home, or long-term care environment [12]	Hospices can have different policies regarding caregiver availability (most—but not all— require someone available 24/7). Some may also allow a trial at home in the setting of a recent symptom exacerbation if the patient's primary wish is to be in the home, or a home-like setting, and there is a plan in place to address this symptom across care settings	Home hospice includes visits from the IDT and can include nurses, social workers, chaplains, home health aide, and trained volunteers, with the physician of record overseeing care
Continuous	<i>Continuous care</i> can be provided in times of crisis (must be 8 out of 24 h), and 50% of this time must be provided by a nurse [12]	It is important to discuss with the hospice if a patient transferring from an acute care setting into continuous care may be better served by GIP level of care	It is helpful to review with the hospice of their choice what their policies are surrounding providing continuous care
General inpatient (GIP)	<i>GIP</i> level of care typically requires pain or symptom management in an inpatient facility that cannot be provided in another setting [12, 13]	Hospices will evaluate the symptom burden being considered, what has been done to address it, and potential for future exacerbations and complications	Generally, this is not meant to be the final disposition for the patient as they will be transferred to a less intense level of care once symptoms are stabilized. However, the facility may allow the patient to remain if death is considered imminent
Respite	<i>Respite care</i> is meant to provide temporary relief of caregiver responsibilities and has specific parameters surrounding this benefit [12]	Consider respite care for patients whose caregivers require a temporary "respite" from caregiving responsibilities, or if there are temporary reasons why the patient cannot stay in a particular routine LOC setting, but otherwise does not qualify for GIP level of care and is considered routine. Examples include the caregiver themselves falls ill, or a protracted power outage in the home of an oxygen-dependent patient makes return for a few days impossible	Collaboration with social work and other members of the ED IDT may be helpful. Remember: patients may present from hospice to the ED with goals of care unchanged related to social reasons and respite care may provide a solution

working relationships for protocols between the hospital and the hospice of a patient's choice when the need for hospice is identified by an emergency provider, and communication is required to jump-start the transition from the ED to hospice placement. In these cases, a PC consult may not always be required, but through a pre-existing collaboration, the process can be developed to make a referral seamless and routine, rather than requiring admission for disposition determination, especially if not desired by the patient and family.

Case Studies

Case Study 1

Mr. M is a 73-year-old man with past medical history significant for hypertension, coronary artery disease, and stage IV non-small cell lung cancer (NSLC) diagnosed 7 months ago and known metastases to the bone and liver. He has recently developed recurrent malignant pleural effusions, and due to his worsening dyspnea, pleurodesis has been discussed versus a tunneled indwelling pigtail drainage catheter. He presents to the ED today complaining of profound dyspnea, especially with mild exertion such as trying to cross a room from bedroom to bathroom, and he had a small amount of hemoptysis earlier, of about a teaspoon in volume, 4 or 5 times. His last available documentation in your healthcare system's medical record reveals initial discussions were held about hospice with decision to maintain code status as a "full code." He had been awaiting review of potential clinical trials at a nearby academic center before electing to enroll in hospice care. You note that he has been to the ED once already this month and received an emergent thoracentesis and was evaluated by radiation oncologic without plans for further radiation at this time. He also describes worsening tolerance to activities of daily living (ADLs), requiring his wife's help to bathe and dress, and that he is spending more than half the day laying down or sitting in a chair.

Question: Does Mr. M Meet Criteria for Hospice Enrollment?

Gather Information Patients may be referred for hospice services if their illness physiologically meets criteria of a prognosis of 6 months of less, should the illness run its usual course, and, after counseling regarding the implications of this prognosis and available options, have accepted this prognosis and elect a treatment regimen and plan of care which focuses only on comfort-based treatments and services. In this case, documentation by the oncologist supports the introduction of hospice due to the extent of progression of Mr. M's NSLC and specifically mentions that no further disease-modifying treatments are available outside of a clinical trial.

In addition, his worsening performance status also supports this assessment. The use of the Palliative Performance Scale (PPS) may support the evaluation as well. A recent systematic review by Baik et al. assessed the capacity for the PPS—a prognostic index independent of a specific disease, which is based on functional status variables—to estimate survival at the end of life found: "All nine studies reported a significant association between PPS scores and survival among palliative care patients with advanced cancer" [21].

Pause and Consider: How Would You Explore Goals of Care with Mr. M?

Gather Information If needed, consider accessing free resources to guide goals of care discussions, including those available through VitalTalk.org and resources from CAPC (if available at your institution).

Back to the Case Mr. M tells you he has been giving thought to hospice and asks if he will "go there to die." You review that hospice is a Medicare benefit based upon a philosophy of care rather than being in a specific place and that care and services can be provided in many settings, including the patient's home. You review that benefits include careful attention to symptom management with the ideals of maintaining comfort and dignity at the end of life. You also mention that the patient can choose to leave hospice at any time and reenter hospice care at a later date [5, 10]. He states he is tired of repeated visits to the hospital and worries about the stress on his wife. He also notes the clinical trial "doesn't seem like it would help me much anyway." You provide reflective listening and inquire how best to preserve his quality of life for as long as possible. He identifies goals of maintaining independence and being home with family. You echo how his goals can be incorporated into the plan of care and review how hospice can insure these remain cornerstones of his care plan. He and his wife agree with referral to social work and case management to evaluate which hospice would best meet their needs and arrange next steps. You also offer to call his oncologist and provide the update about the hospice referral, and he appreciates this coordination.

Pause and Consider: What Further Information Would Be Helpful to Know for His Disposition and Management?

Gather Information Please reference Tables 61.2 and 61.3 for guidance on what information would be helpful in assessing hospice level of care (LOC) and arranging the enrollment process. A detailed outline of this process is also presented in the article by Drs. Quest and Lamba: Hospice Care and the Emergency Department: Rules, Regulations, and Referrals [1].

Additional Interventions During your phone call with the oncologist, they agree with the hospice plan and elect to be the physician of record for Mr. M's hospice care. You share this news with Mr. M who is appreciative of this continuity of care and notes he has provided the choice of hospice to the social worker on the case. After making the referral, the social worker asks you to report the symptoms Mr. M is experiencing and his current management for the hospice to review. After reviewing his vital signs, you note that he is mildly hypoxic at 93% on RA at time of arrival and has improved to 97-100% on oxygen via nasal cannula. He is otherwise hemodynamically stable, and his chest X-ray reveals adequate placement of the pigtail catheter; however he has had worsening evidence of possible lymphangitic spread from imaging done on his last visit to the ED and continues to complain of significant subjective dyspnea, despite improvement in his oxygen saturations. Based on his volume status (which you assess as likely hypovolemic), you opt to avoid diuresis at this time and instead implement lowdose oral morphine, as opioids can alleviate the sensation of subjective dyspnea [22]. Mr. M does not identify symptoms aside from dyspnea and reviews his priority of returning home again on reassessment. You broach code status with him, and after careful counseling, he elects to remain full code as he is looking forward to seeing his grandson graduate from college in 2 weeks and demonstrates in his responses that his code status is emotionally tied to this milestone event.

Consider: What Might Be Important to Sign Out to the Hospice of His Choice?

Gather Information Review what you would typically sign out for ongoing care for a patient, and consider how this care should be transitioned across settings.

In your review of his overall clinical status, including hemodynamic stability and improvements in subjective dyspnea with low-dose opioids, your assessment concludes that his comfort can be maintained at home with routine hospice care as his symptoms likely do not require rapid medication titration. If Mr. M did require more aggressive control of his symptoms, and more frequent reassessment and titration, these could be available in an IPU or via short-term, continuous care services in the community. Returning to his own home is consistent with his goals of care, and social work reveals that he has 24/7 caretaker support in the home. Further, the hospice is able to arrange for a nurse liaison who can do an intake visit tonight, with DME equipment delivery tomorrow morning, which is acceptable to the patient. You confirm this with the hospice, review his decision to remain full code at this time, and also discuss that he has a pigtail catheter that he has previously been given education on utilizing and that he will require ongoing nursing care to maintain. You also request oxygen for symptomatic control of dyspnea, with the standard hospice parameters of "titrate for comfort." After summarizing his case, the hospice provides instruction and requests prescriptions for medications to address symptoms that may evolve in the coming days.

Case Study 2

Ms. H is a 90-year-old woman with history of Alzheimer's dementia presenting from home hospice to the ED via EMS. Her vital signs are unremarkable, and her daughter reports that she is concerned hospice is not providing the care that her mom requires. After further discussion, you learn the patient's daughter was recently diagnosed with breast cancer and requires surgical intervention with a lumpectomy and lacks other social supports to continue caring for her mother at home. The patient's physical exam is benign overall, and the daughter denies changes in behavior or new symptoms. Your exam is consistent with an alert, non-verbal patient who does not follow commands or provide responses to questions but does not appear to be in any acute distress. You do not observe signs of neglect. Your colleague inquires if you're planning to admit the patient given limited resources at home for social reasons?

Question: What Next Steps Should Be Considered?

Gather Information

- Confirm goals of care based on substituted judgment.
- Evaluate the family's concerns and underlying reasons for presentation.
- Consider how collaboration with the patient's hospice can be utilized.

You call the patient's hospice and learn that the hospice interdisciplinary team (IDT) was unaware of the daughter's diagnosis and impending surgery. They report the patient has only been on service for 5 weeks, and the IDT has found the daughter to be very private about sharing personal information and is unmarried with no children to offer additional support. In addition, the IDT shares that she has appeared reluctant to accept assistance at times, including declining the use of care aides and volunteers. After learning this information, the hospice offers to send a social worker to the ED to discuss care plan options and goals of care directly with the daughter. Via the social worker, the hospice offers to provide 5 days of respite care at a local skilled nursing facility, and after further discussion with the patient's daughter, she agrees to this plan and is appreciative. The hospice social worker also shares with you that the daughter is the youngest of the patient's five children and the only daughter. She has voiced her fear that her brothers (all out of state) will feel she is not taking "good enough care of mom" if she allows others to help or to allow a respite. The social worker and the patient's daughter telephoned two siblings from the ED who both confirm that all the siblings consider the daughter to be the spokesperson for the family with plans to inform the remaining brothers of the plan. They provide reassurance to the patient's daughter and endorsed the need for the daughter to take care of herself and for the respite to proceed.

This case underlines the importance of collaboration with the hospice agency when a patient presents to the ED while enrolled in hospice services. It is necessary to have careful discussions with the patient's surrogate decision-maker for healthcare to uncover the reason behind the presentation and confirm goals of care. At times, addressing communication challenges between the patient-family unit and the hospice is essential to providing the care the patient needs and can avoid unnecessary admissions.

Case Study 3

Mr. R is a 68-year-old man with history of COPD on 2L of oxygen via nasal cannula at home and long-term steroid

dependency. He presents with a femoral neck fracture acutely after tripping over his oxygen tubing. After initial stabilization and consult to orthopedics, you note that he has recently had repeated exacerbations requiring frequent hospitalization (three times in the last 2 months) and his FEV1 is less than 35% of predicted with weight loss of over 15% in the last year as noted by his pulmonologist. He also has had worsening functional status with recent need for a move to an assisted living facility. His daughter is a physician and advocated for him to be a lung transplant candidate recently with reported decision that he was not a candidate at a major academic center during the last month. He also has not regained function with pulmonary rehab and has been determined to not be a candidate for surgical intervention. In essence, his pulmonology records indicate that they are hoping to achieve stability with his new, worsened, clinical status. You assess that he is demoralized and asks you if there is an alternative to returning repeatedly to the hospital?

Pause and Consider: What Factors About His Presentation Make Him a Candidate for Hospice? If He Elects Hospice, Is He Ineligible for Operative Management of His Fracture?

Assume the Following Has Been Pursued

- You consult to hospitalist service, and they plan to admit him and further review his pre-operative risk.
- A femoral nerve block significantly improves his pain.

Gather Information You confirm that his FEV1, ongoing oxygen requirements, repeated admission, and need for ongoing steroids and other pulmonary treatments are considered indicative of a more guarded prognosis [23]. Furthermore, his lack of other options to manage his disease state further supports a hospice admission.

Patients can revoke hospice due to both changing their mind about goals of care (including when a new treatment becomes available) and be admitted for procedures that are considered to have greater benefit than burden for their quality of life based on their goals of care [12]. An unstable femoral neck fracture in an ambulatory patient with a prognosis of months to years is likely to have significant impact on Mr. R's quality of life, and operative intervention (if he is considered a candidate) may allow him to have greater mobility and less pain over the time that he has remaining. In addition, COPD is historically difficult to prognosticate, and the factors mentioned above generally have been used to assess prognosis in terms of years rather than months [23]. B. K. Marr and D. J. Doberman

Consider palliative medicine consultation in addition to assessment of pre-operative risk and orthopedics consultation to help review possible treatment options, confirm goals of care, and promote shared decision-making.

Conclusion

Hospice offers an important resource for patients reaching the end of their lives and their families. Emergency physicians should be skilled at identifying patients eligible for hospice services and addressing this topic as part of a goals of care discussion, as well as assessing the unmet needs of hospice patients who present to the ED from a setting where hospice care is being delivered [24, 25]. Cancer represents a disease where hospice may eventually become appropriate based on the patient's goals and overall clinical status. Therefore, just as it is necessary to recognize common oncologic emergencies in order to provide excellent care, recognizing hospice eligibility, exploring reasons for presentation from hospice settings, confirming goals of care, utilizing resources within ED IDT and PC teams, and collaborating with hospice teams in providing care across settings are all important skills for the emergency provider to have at the ready for this patient population with complex care needs. This chapter has sought to review the basic steps in this process and underlines how the ED is a critical setting for assessing the needs of patients at pivotal points in their disease trajectory and offers an important opportunity to insure the delivery of ongoing goal-concordant care.

References

- Lamba S, Quest TE. Hospice care and the emergency department: rules, regulations, and referrals. Ann Emerg Med. 2011;57(3):282–90.
- 2. Chan GK. End-of-life models and emergency department care. Acad Emerg Med. 2004;11(1):79–86.
- National Hospice and Palliative Care Organization. National hospice and palliative care organization facts and figures 2018. https://www.nhpco.org/wp-content/uploads/2019/07/2018_NHPCO_ Facts_Figures.pdf. Accessed 11 Aug 2020.
- Meier DE, Morgan L. Center to advance palliative care. Messaging palliative care. [updated February 21, 2019] https://www.capc.org/ blog/palliative-pulse-palliative-pulse-november-2016-messagingpalliative-care/. Accessed 11 Aug 2020.
- Centers to Advance Palliative Care. Patient and family resources. https://www.capc.org/about/patient-and-family-resources/. Accessed 11 Aug 2020.
- National Institute on Aging. What are palliative care and hospice care? https://www.nia.nih.gov/health/what-are-palliative-care-andhospice-care#hospice. Accessed 11 Aug 2020.
- Medicare.gov. Hospice care. https://www.medicare.gov/coverage/ hospice-care. Accessed 11 Aug 2020.

- Grudzen CR, Richardson LD, Morrison M, Cho E, Morrison RS. Palliative care needs of seriously ill, older adults presenting to the emergency department. Acad Emerg Med. 2010;17(11):1253–7.
- Lamba S, Quest TE, Weissman DE. Emergency department management of hospice patients #246. J Palliat Med. 2011;14(12): 1345–6.
- Lamba S, Quest TE, Weissman DE. Initiating a hospice referral from the emergency department #247. J Palliat Med. 2011;14(12):1346–7.
- Jackson VA, Jacobsen J, Greer JA, Pirl WF, Temel JS, Back AL. The cultivation of prognostic awareness through the provision of early palliative care in the ambulatory setting: a communication guide. J Palliat Med. 2013;16(8):894–900.
- Lamba S, DeSandre PL, Todd KH, Bryant EN, Chan GK, Grudzen CR, Improving Palliative Care in Emergency Medicine Board, et al. Integration of palliative care into emergency medicine: the Improving Palliative Care in Emergency Medicine (IPAL-EM) collaboration. J Emerg Med. 2014;46(2):264–70.
- Center to Advance Palliative Care. Integrating palliative care practices in the emergency department. https://www.capc.org/ toolkits/integrating-palliative-care-practices-in-the-emergencydepartment/. Accessed 11 Aug 2020.
- George N, Barrett N, McPeake L, Goett R, Anderson K, Baird J. Content validation of a novel screening tool to identify emergency department patients with significant palliative care needs. Acad Emerg Med. 2015;22(7):823–37.
- George N, Phillips E, Zaurova M, Song C, Lamba S, Grudzen C. Palliative care screening and assessment in the emergency department: a systematic review. J Pain Symptom Manage. 2016;51(1):108–19, e2.
- Glajchen M, Lawson R, Homel P, Desandre P, Todd KH. A rapid two-stage screening protocol for palliative care in the emergency

department: a quality improvement initiative. J Pain Symptom Manag. 2011;42(5):657-62.

- Penrod JD, Deb P, Dellenbaugh C, Burgess JF Jr, Zhu CW, Christiansen CL, et al. Hospital-based palliative care consultation: effects on hospital cost. J Palliat Med. 2010;13(8):973–9.
- Meier DE, Beresford L. Fast response is key to partnering with the emergency department. J Palliat Med. 2007;10(3):641–5.
- Penrod JD, Deb P, Luhrs C, Dellenbaugh C, Zhu CW, Hochman T, et al. Cost and utilization outcomes of patients receiving hospitalbased palliative care consultation. J Palliat Med. 2006;9(4):855–60. Erratum in: J Palliat Med. 2006;9(6):1509.
- Highet BH, Hsieh YH, Smith TJ. A pilot trial to increase hospice enrollment in an inner city, academic emergency department. J Emerg Med. 2016;51(2):106–13.
- 21. Baik D, Russell D, Jordan L, Dooley F, Bowles KH, Masterson Creber RM. Using the palliative performance scale to estimate survival for patients at the end of life: a systematic review of the literature. J Palliat Med. 2018;21(11):1651–61.
- Vargas-Bermúdez A, Cardenal F, Porta-Sales J. Opioids for the management of dyspnea in cancer patients: evidence of the last 15 years–a systematic review. J Pain Palliat Care Pharmacother. 2015;29(4):341–52.
- Wilson Childers, Julie, Arnold, Robert, Curtis, J. Randall. Fast facts and concepts #141 prognosis in end-stage COPD. Palliative care network of Wisconsin, fast facts. https://www.mypcnow.org/fastfact/prognosis-in-end-stage-copd/. Accessed 7 Jul 2020.
- 24. Philip J, Remedios C, Breen S, Weiland T, Willenberg L, Boughey M, et al. The experiences of patients with advanced cancer and caregivers presenting to emergency departments: a qualitative study. Palliat Med. 2018;32(2):439–46.
- Beemath A, Zalenski RJ. Palliative emergency medicine: resuscitating comfort care? Ann Emerg Med. 2009;54(1):103–5.



Communication

Bonnie K. Marr, Kate Aberger, and Rebecca Goett

Introduction

This chapter will focus on how we communicate with the oncologic patient and their families in the emergency department (ED). Clinicians will learn how to facilitate discussion of emotionally charged topics with patients in a compassionate, effective, and timely way.

It can be argued that communication is as important as the technical aspects of treating disease [1-5]. Patients are suffering and they come to us for help. As physicians, we seek to understand the person with the disease and their wishes for the management of what ails them [6-8]. We can learn to harness our empathy – detecting another's emotions and experiencing their feelings of suffering – and learn to respond to this suffering with compassion. This type of communication can have positive measurable outcomes, for both the patients and us [9-13].

The main skill that palliative physicians practice frequently is a *goals of care* conversation. Although lengthy care discussions may seem impractical in the ED, there are methods to help acute care providers engage patients in these difficult conversations in brief but effective ways. We will use cases to illustrate "abbreviated goals of care conversations" [14] and give the reader some concrete examples of techniques to incorporate into everyday practice.

Components of an initial assessment by the emergency physician not only are vital to the provider determining a differential diagnosis and treatment plan but are also essential elements to consider in reviewing options with the patient

B. K. Marr (🖂)

Section of Palliative Medicine, Johns Hopkins Hospital, Baltimore, MD, USA e-mail: bmarr1@jhmi.edu

K. Aberger

R. Goett

and family in shared decision-making. As we are doing our initial resuscitation, we ask ourselves: Is this a reversible process? Is this a treatable process (but one that will not reverse the underlying disease), or is this patient actively dying [15]? Having transparent dialogue with a patient or family and building a shared plan for ongoing management help ensure clear communication about what next steps and expected outcomes may be in order to help patients and families feel prepared. Frequently the emergency physician has mere minutes to build rapport with a patient and gather information about his/her preferences for care. How do we identify those in need of deeper conversations? How do we recognize patients with unmet palliative care/communication needs? General guidance exists for emergency providers regarding when palliative care consultation and interventions should be considered. George et al. has developed a screening tool (Fig. 62.1), and the Centers to Advance Palliative Care has an initiative called "Improving Palliative Care in Emergency Medicine" or IPAL-EM, which also offers a screening tool [16–18].

Prognosis

In order to recognize a worsening prognosis, we must first look at the trajectory of the patient's illness (Fig. 62.2) [19].

Prognosis is generally based on two variables: (1) disease state and (2) functional status. In the oncologic patient, there are a few guidelines that we can follow. As shown above, terminal illnesses such as cancer usually preserve functional status until near the end. Then a precipitous acceleration of illness occurs once functional status begins to decline. Therefore, functional status is key to prognosticating in cancer patients. The median survival times for multiple brain metastasis, malignant hypercalcemia, malignant pericardial or pleural effusion, carcinomatous meningitis, malignant ascites, and malignant bowel obstruction are relatively short (weeks to months) and can be estimated with some degree of accuracy [15].

Division of Palliative and Geriatric Medicine, Department of Internal Medicine, St. Joseph's Health, Paterson, NJ, USA

Department of Emergency Medicine, Rutgers New Jersey Medical School, Newark, NJ, USA

K. H. Todd et al. (eds.), Oncologic Emergency Medicine, https://doi.org/10.1007/978-3-030-67123-5_62

Fig. 62.1 Palliative care screening tool. (From George et al. [16], with permission Springer Nature)



Regardless of medical specialty, discussing prognosis with the patient is a challenge. The emotional and psychological impact of the information imparted and the lack of precision possible under most circumstances make it fraught [18]. For this reason, it is important to ask permission prior to sharing this information. If a patient or family is focused on prognosis, it is often useful to ask: "Why are you asking me about this now?" This will usually lead to the patient or family talking about their anxiety or fear, which you can then address. We provide prognosis in terms of "hours-to-days," or "days-to- weeks," or "weeks-to-months," or "months-toyears," with the provision that these are estimates. Using Fig. 62.2 Proposed trajectories of dying. (From Lunney et al. [19], with permission John Wiley & Sons)



specific numbers of hours, days, weeks, months, or years leaves more room for error and introduces potentially emotionally traumatic sequelae if the patient were not to live as long as forecast. Conversely, some patients and families will be upset if the patient lives beyond that estimation and perhaps express confusion and lack of faith in medical information.

When you find that further conversation is needed, you must set the stage and prepare. The emergency physician should facilitate communication among the various multidisciplinary teams involved in the patient's care. Palliative care teams, where available, can also assist in complex decision-making. Consider utilizing pre-existing decision tools to determine if a palliative care consult may be helpful [16–18].

The Building Blocks of Effective Communication

Preparation

Good communication begins prior to entering the room. Gather as much information as possible. Examine the medical record carefully. Make contact with the treating oncologist or primary doctor. Tell them what you have found. Ask what their goals are for the patient, and discuss your plan to discuss your findings with the patient. Tell the specialist your concerns and ask for their recommendation. You can tell your colleague the exact words you plan to use: for instance, "I am worried this cancer is progressing, I want to talk with you about the possibility of things not going the way we had hoped."

Capacity

Assess the patient's capacity. Medical decision-making capacity is the patient's ability to understand the benefits and risks of, and the alternatives to, a proposed treatment or intervention (including no treatment) [20–22]. Capacity is the basis of informed consent. If you feel the patient does have capacity, prepare them by asking permission to have a "difficult conversation." Tell them it will be helpful to have someone with them either in person or on the phone. It is generally recommended not to speak to the patient alone unless you have capacity, or you are worried that they may be too emotionally burdened to have the conversation, find the legal next of kin.

Physical Space

Be sure that you have a quiet, separate place to have the conversation. Try to make it as private and comfortable as

Table 62.1	The REMAP Guide
Reframe	What is your understanding of what the doctors have told you about your dad's health? If their understanding differs from the clinical realities, consider a follow-up statement such as Unfortunately, we are in a different place now. Is it okay if we talk more about what we are seeing and what the next steps may be?
Expect emotion	Stop talking and listen! Consider responding to emotion directly. Examples include the following: <i>I can see you</i> <i>are really concerned about this. Is it okay if we talk more</i> <i>about what this means</i> ? Or <i>I cannot imagine what it has</i> <i>been like to hear all this news.</i> Do not be afraid of silence
Map out the future	Try and identify what the patient's goals are before making any recommendations. <i>Given what we have just</i> <i>discussed about the illness your dad is dealing with right</i> <i>now, what do you think would be most important to him</i> <i>right now?</i> Or <i>thinking about what you dad put in his</i> <i>advanced directive and your knowledge of him as his</i> <i>loved one, is there anything he would want to avoid?</i> <i>What would his reasons be? What values in life are most</i> <i>important to him?</i>
Align with values	Now that I have a better understanding of what is important to your father, let's talk more about treatment options
Propose a plan	Give specific recommendations, rather than a menu of options. <i>From what you have told me, I recommend How does that sound to you?</i>

Adapted with permission from VitalTalk O Market https://www.vital-talk.org/guides/transitionsgoals-of-care [23]

possible. Have tissues available. Have a planned strategy for how the conversation should unfold.

Take a moment, ground yourself, clear your mind of any prior encounters, and rid yourself of preconceived ideas. Make a conscious effort to focus on being an advocate for the patient and meeting them where they are in the journey of serious illness. Also, body language speaks volumes. Walk in, make eye contact, and introduce yourself. Sit down if possible, or if not, assume a pose that says you are not rushed. Allow the patient to speak for the first minute without interrupting.

Conversation Templates

Most of these difficult conversations follow a template, and steps can be tailored to the patient depending on their immediate need (Tables 62.1 [23] and 62.2 [14]).

Responding

Do we flinch internally when we hear a family say, "I want everything done" or "He is a fighter"? We must learn to rec
 Table 62.2
 Approaching 5-min ED goals of care conversations systematically as a procedure

Phase	Action
Minutes	Elicit patient understanding of underlying illness and
1-2	today's acute change
	If available, build on previous advance directives or
	documented conversations
	Acquire sense of patient's values and character (to help
	frame prognosis and priorities for intervention)
	Name and validate observed goals, hopes, fears, and
	expectations
Minutes	Discuss treatment options, using reflected language
3–4	Continually re-center on patient's (not family's) wishes
	and values
	Recommend a course of action, avoiding impartiality
	when prognosis is dire
Minute 5	Summarize and discuss next steps
	Introduce ancillary ED resources (e.g., hospice/
	observation unit, social work, chaplain)

From Wang [14], with permission Elsevier

ognize these statements as possible emotional responses rather than cognitive responses (Table 62.3) [24]. Instead of this putting an end to the conversation, we should explore what "everything" means to this particular family.

If we can respond to their emotions with inquiry and recognition instead of continuing to give medical information, we can open up dialogue in a way that will help the family process what is happening emotionally. Once their emotions are realized, often discussions about care can move forward. If not, best to continue resuscitative efforts as appropriate, and prepare the family for further discussions by the inpatient or palliative care teams.

Effective Use of Silence

Sitting in silence may feel as though it goes against the grain of a provider's role in a dynamic conversation. Physicians may infer that a pause in the conversation means that there should be further explanation or that more active direction is needed. However, the use of silence and pauses in the conversation aids communication. Silence allows the patient or family time to process information, cope with associated emotions, and utilize the time to formulate next questions or communicate concerns. If a provider jumps in too soon to fill this proverbial "space," the patient and/or family may not have the opportunity to move through these steps, if needed. After a pause, it can be helpful to incorporate a "check-in" and offer to repeat any needed pieces of information. It is appropriate to also acknowledge the difficulty of absorbing the information proffered and to offer support.

But he was doing great	Medical	He is now in septic and cardiac shock
when I visited him last!	Emotional	I cannot imagine what a shock this is for you. Unfortunately he is very sick. He has a very serious infection and his heart is failing. We are all hoping for a different outcome, but we are worried that he is dying
Do everything possible!	Medical	Ok, we will do CPR, intubation, dialysis, and any other procedure
	Emotional	I hear you, and we will provide him with the most aggressive care. It sounds like this has been a hard/long journey and this is a terrifying time. Can I ask, has your father ever imagined his health worsening to what we see now?
He needs a feeding tube	Medical	We know from studies that feeding tubes do not help in the end of life
for nutrition so he can get stronger!	Emotional	Most people think of food as love. We think it is inhumane to withhold food from people we love. But your father's body is shutting down from his illness, and he does not feel hunger like he used to. Artificial nutrition is not the same as food and will not give him strength. A feeding tube in his stomach may deprive him of the joy of tasting his favorite foods. Can we talk instead about ways to increase his pleasure around the food he may want?
You want me to pull the	Medical	We will continue to do everything
plug? You want me to stop withdraw care? You want me to let him die?	Emotional	I hear your concern. It feels like we are making this decision, but really it is his body that has decided. He is dying a natural death from this illness. I will do everything I can to aggressively make sure your father does not suffer as his body goes through the natural process of dying

 Table 62.3
 Responses to patients' and caregivers' reactions)

Adapted from Aberger and Wang [24], with permission Springer Nature

Language, Diversity, and Culture

These intense conversations are difficult in any language. If the patient only speaks a language you do not, it adds another layer of complexity. We find, of course, that live interpreters are best, if available. Family members should not be used for these sensitive conversations because often, as family members, they are subject to the emotional content of the information and might not be able to translate the conversation. If language line phones are the only option, it helps to prepare the interpreter for the type of discussion you are going to have as they can also become discomforted when discussing these sensitive topics, and it may help to organize the conversation.

Cultural considerations should always be explored. We find that no one is an expert on all cultures, but sensitivity and curiosity are essential. When in doubt, ask. Ask openly about custom or tradition. Apologize if you find you have overstepped or assumed incorrectly. Some cultures cannot talk about death directly, so exploring which words are acceptable may take an extra few minutes. This is essential in establishing trust and rapport.

Spiritual and Religious Considerations

Spiritual and religious needs are also diverse and should be explored. Here again, curiosity and sensitivity are key. Ask about needs, beliefs, and customs. Recognizing spiritual distress is also very useful in having these conversations. Often families will respond to our bad news or recommendations with the phrase, "We are hoping for a miracle." This is another phrase that may invoke our discomfort. If we can

 Table 62.4
 AMEN communication tool for miracles [24, 25]

Affirm/Meet and Reflect	Demonstrate respect and communicate empathy for what the patient/family values and wants
Learn more	Get curious and ask to hear more about the miracle, hopes, or values framed through the patient/family religious/cultural beliefs
Broaden the care plan	Determine what else is important so you can discuss different care plan options from the perspective of patient/family values and goals
Educate	Educate about the medical situation and options
No matter what	Reassure the family of your support no matter what happens

recognize and manage our own reactions, we can use the following communication protocol to open this statement up, thus broadening and continuing our discussion. It is called the AMEN protocol (Table 62.4): "Affirm, Meet, Educate, No matter what" [24, 25]. This is a powerful tool to keep the conversation going rather than continuing attempts to "solve an issue."

Case Studies: Starting with Common Dilemmas and Gradually Layering in Complexity

Case Study 1: The Rapidly/Actively Dying Patient

The patient is a 62-year-old male with metastatic lung cancer who presents with altered mental status. Labs show pancytopenia with a platelet count of 5. He becomes more lethargic and requires intubation. His pupils are unequal and you suspect an intracranial hemorrhage.

Approach to Goals of Care for the Actively Dying Patient

If the patient is declining rapidly, goals of care are needed quickly. Think briefly about the trajectory of their illness and where they fall on it. This will lead to an initial prognosis. We can use this prognosis to guide our discussions – if prognosis is hours to days, we likely need to start conversations now in the ED. Even if we proceed with full life support resuscitative measures, we need to initiate discussions about the strong possibility of further decline or death.

A useful mnemonic is ABCD [18]. We need to consider this when the patient is chronically ill, actively dying, or at the end stages of their illness.

- A: Advanced directives: Does this patient have any? POLST, MOLST, prior PC involvement?
- B: Better symptom management
- C: Caregiver? Who is the next of kin?
- D: Is the patient decisional?

You identify the family, including the next of kin. They state their loved one has been declining and he expressed the desire that "when my time comes, let me go." They are tearful but agree with your recommendation based on prognosis, and his past expressed wishes to the family for comfort measures only.

Communicating with the Family of an Actively Dying Patient

Studies confirm the importance of the explicit explanation of impending death signs and the importance of avoiding both vague explanations about future changes and excessive warning of sudden change [13, 26]. Research tells us that generally families also want to know a patient's relief of suffering, advice on how to care for the patient, allowance of time for the family to grieve, and that providers take care to ensure that family members cannot overhear conversations between the medical staff outside the room at the time of patient death [27].

Families will often have questions about what to expect – whether the patient can hear or is aware of their presence and if the patient is suffering due to a lack of nutrition or hydration. It is important to answer all these questions and concerns. We encourage families to spend time and speak to the patient with acknowledgement that the patient may hear or sense their presence. The patient's religious and cultural traditions may also offer resources. Social work and pastoral care also should be considered.

Case Study 2: Stable Patient with Progression of Disease

Breaking Bad News Worsening disease despite treatment [28, 29]

The patient is a 64-year-old male with past medical history of hypertension and colon cancer post resection with chemotherapy, years ago. He presents to the ED with cough and shortness of breath. He has no hemoptysis or fever but is tachycardic. You do a CT scan to rule out pulmonary embolism. No embolism is found, but multiple pulmonary nodules are seen, suspicious for metastatic disease.

Before You Go in the Room Preparation is key. Do a thorough chart check, or call his primary care provider or oncologist to confirm this is possibly a new disease progression. If you find no evidence showing prior metastatic disease, you have to discuss the results with the patient.

In the Room Make sure the patient has capacity and has support present or by phone. First ask broadly what they know about their cancer. As stated in the previous case, asking this sets where to begin the conversation [17, 23].

Some patients have an understanding that their cancer has returned or spread to other parts of their bodies. Other patients and family may believe the cancer is "cured" or in remission, or that the cancer has spread, but they believe that current treatment is curative rather than palliative. If they disclose this, you can ask about treatment. What do they hope to get out of their treatment?

Once you know where to begin the conversation, discuss the results simply. Often begin with a "headliner" [23] that is concise and gets the critical information across such as "I have the results from the CT, it looks like the cancer might have spread/returned/is getting worse." Using a VitalTalk "I'm worried…" statement [23] here can also be effective in conveying the results and your sympathy. Although you often need to give more information after this statement, pause for the patient's response.

Next, Gauge Where the Patient Is Emotionally

If there is tension or silence, you may need to circle back to discuss future steps. Tell the patient you will return. Alternately, if the patient immediately and rapidly begins asking questions, this may be part of an emotional response to the news. He may not want further medical details, but fear and anxiety are driving him to ask. Name the emotion you are seeing.

When you gauge that he is ready to discuss disposition and plan, start by recognizing the difficult nature of the conversation: "Whenever you are ready, I'd like to discuss what's going to happen next" [23]. In addition, offer palliative and/or spiritual care services, explaining what they do and the essential support they can give in the difficult days/ months ahead.

After Discussion

If the patient is not actively dying and is stable for transport, it may also be appropriate to consider transfer to a hospice inpatient unit or home hospice. Provide guidance to the family that it is possible the patient may die during transport with more specific information tailored to the individual patient's clinical condition and estimated distance/time in transport. Some family members may elect against transport for these reasons.

Case Study 3: Death Disclosure of Sudden and/ or Unexpected Death

You receive a call from paramedics reporting a cardiac arrest in the field. On arrival, emergency medical service (EMS) providers state they found the 73-year-old male after an unwitnessed cardiac arrest at home. He was undergoing chemotherapy for prostate cancer. EMS states that the son found the patient not breathing this morning and called 911. EMS intubated him and began ACLS protocol with pulseless electrical activity on the monitor for 20 min. In the ED, you continued CPR for over 15 min without a return of pulses, and the monitor now reveals asystole. Your team pronounces the patient in the ED. The charge nurse informs you that the family has arrived and is in the waiting room.

Before You Go in a Room

Establish your Team. The ED is hectic; however, nurses, residents, and other staff involved in resuscitation often want to participate in family meetings. Calling outside staff that can assist with bereavement (e.g., spiritual care, chaplains, or counselors) may also be helpful when available. Discuss beforehand which team members will likely stay with family afterward for instructions or family support.

Set the Room. Quiet is often hard to come by in the ED, but a family room is ideal. Social worker's or nurse manager's offices can also suffice, depending on the size of the family. Have one of your team members or staff escort the family to a quiet place. The person placing family in the room should ideally not be you in order to avoid questions or forcing you to disclose news in the hallway or in front of other patients. Also, this staff member can identify a "head" of the family or next of kin. Staff can ask that designated loved one who they think should also be in the room when the meeting begins. In addition, this co-worker can report back to you who and how many family/friends there are and their current emotional state.

In the Room [23, 30]

Introduce. Introduce yourself to every person in the room. Find the lead family member and place yourself next to them; this will be the person you primarily interact with. First, ask what they know so far, for example, the son watching CPR in progress has a better understanding of the severity of their loved one than the son called at work by the nursing home stating, "Your family member was transferred to the ED." Asking this establishes where to begin the conversation.

Deliver. Then give a "warning shot" stating you have difficult news. Very briefly without jargon, describe what happened today. Usually the events can be summarized in no more than two to four sentences. When disclosing a death, use the word "died" and then pause for silence. Traumatic, sudden, or pediatric deaths often produce more raw emotional responses. These will be discussed in more detail shortly.

Comfort. Find phrases that you feel comfortable saying during this difficult conversation such as "I wish this hadn't happened," "I'm sorry for your loss," or "I can't imagine how you feel." Some physicians feel their expressions of sympathy sound contrived or even condescending. However, empathy can facilitate further dialogue with the bereaved and is usually appreciated.

After Discussion

Assist. After allowing loved ones to process the event, ask for any questions. Discuss next steps and hand off the conversation to one of your colleagues. Social workers or case managers can obtain more information about the patient. Nurses or patient assistants can accompany them to their loved one for viewing. Special instructions are needed for traumatic deaths, such as no touching the deceased. Sometimes law enforcement may need to further discuss the death with medical providers.

Hand Off. Leaving the room, especially when it is still full of emotional people, can be awkward. Introduce the staff that will stay behind, and ensure the team is available for any questions. If you are in a family room, you can offer the space to the bereaved to take their time to contact family, and delegate and discuss who and when (or if) a family member will see the deceased.

Case Study 4: Managing Requests to Withhold Information

Your patient is a 78-year-old female admitted from home with a chief complaint of shortness of breath and not eating for the past 2 weeks. Past medical history includes breast cancer s/p chemo, stopped 2 months ago due to weight loss and weakness. Patient is alert and oriented to person, place, and time. She says she is not hungry and just tired all the time. Her chest X-ray shows worsening metastatic disease in lungs.

You discuss the results first with her daughter, her primary caregiver and health proxy. Her daughter asks information not be disclosed to the patient.

Requests for non-disclosure can be complex and often reflect a family's member's efforts to protect the patient from emotional harm. The flow of information may have been agreed upon within the family already (e.g., the patient does not want to know), but we must explore this prior to making assumptions.

The following modified approach is proposed by Chaitin from the Palliative Care Network of Wisconsin [31]:

Before You Respond Remain calm and monitor your inner reactions. Explore the request for non-disclosure with curiosity. For clinicians this can be frustrating, but expressing how *you* feel can quickly end the conversation. Recognize and check your emotions. Remember, the calmer you remain, the more information you will obtain from the family as to why they do not want their loved one to be informed.

Gain Knowledge

Psychosocial and Cultural Aspects. Use open-ended questions to gain insight. "Can you tell me more about your request?" "How does your loved one [your family] typically handle difficult information?

The Disease Itself. Inquire what the patient knows about their disease. Does the family think the patient already knows or suspects the results and would rather not talk further about it? Or is the patient completely unaware? Frequently, both the family and patient may be frightened and attempt to protect each other by not talking about "bad news."

Plan. There is no one right way. As the clinician, you can recommend first asking the patient if she would like to know or discuss her health. If so, we prepare the family by reassuring them that we will be gentle and slow. We will share small pieces of information and continue to ask the patient permission to continue.

Ask the patient in a neutral way (e.g., listing options so the patient does not feel pressured to choose only one specific way). Having family present and rehearsing beforehand are also helpful so the family feels they still support the patient. For this case, "Your family brought you to the hospital because you were short of breath and not eating. We took tests to figure out why. Some patients want to know all their tests and results, others prefer the doctors talk with family about what is happening and how to best help them. What do you think?"

Case Study 5: When There Are Disagreements Among Physicians

An 83-year-old male presents with SOB, cough, fatigue, and weakness. PMH includes prostate cancer on chemotherapy. Chest X-ray today reveals worsening metastatic disease with new pleural effusion. In the past year, the patient has had multiple ED visits and admissions with a diagnosis of advanced metastatic disease and a pattern of overall decline. Patient shows little understanding of disease progression.

When questioned about his cancer, the patient states: "I am a fighter, and I'm going to beat it and be cured." When you speak with his oncologist, she feels the patient can live for years, and his chemotherapy overall is going well.

Before You Respond [32–34] When dealing with a response that is discordant, emergency clinicians must first remain calm and try to understand the other clinician's viewpoint. Sometimes in an established provider-patient relationship, there can be discrepancies in prognosis understanding. This can be due to several reasons (e.g., the clinician-patient bond) that make it difficult to acknowledge that the disease is progressing, regardless of treatment. Also sometimes patients can take away or naturally choose to accept only the positives during care discussions despite clinicians delivering information to the contrary.

Gain Knowledge Begin by using open-ended questions to gain insight on how the oncologist or primary clinician is interpreting the patient's status. Ask how past planning and prognosis discussions between him/her and the patient have progressed. Using "I'm worried" statements such as "I'm worried today that he's doing worse" can help "third space" the disease so the oncologist does not feel confronted or feel a sense of failure [23]. It is hoped that the oncologist can reflect on the overall trajectory of the patient's disease and give insight into the current negative results.

Plan State to the oncologist that you will share today's results with the patient and ask him/her: "Based on these results, how can we best help the patient?"

Acknowledge their difficulty in reframing their expectations for the patient such as "Seems like you have

known the patient a long time. This must be difficult for you." If the patient is not to be discharged, tell the oncologist that as the ED clinician, you want to include them in further plans. "Would you mind if I conferred with you in the next steps? From the ED, we are planning to...." In addition, state that you are consulting palliative care, stating they will provide an additional level of support during this admission. If the oncologist disapproves of the plan, remember: it is his/ her opinion. Ultimately your obligation is to the patient, not the oncologist.

Case Study 6: Dying in Isolation

You care for a 63-year-old female with a chief complaint of shortness of breath, cough, and fever. She has a past medical history of COPD and lung cancer. She is alert and oriented, but hypoxic on room air and febrile. You are concerned about a COVID-19 infection. In the ED, the patient becomes more hypoxic and has increasing oxygen demands. She is ultimately intubated and awaits admission to ICU. The patient's family arrive several hours later, but are not allowed in. They beg for information about her condition and ask to see the patient.

Before You Discuss You first have to prepare for your meeting. Communication and goals of care conversations have evolved during COVID-19 due to the unpredictability of the pandemic. Communication between patients, clinicians, and family varies widely. Supporting a patient's loved ones can take place on voice or video calls as visitors are often not allowed. Communication tools are available from CAPC and VitalTalk for periods of social upheaval [23, 35]. These conversations are best led with an effort to first identify patient values. Recommendations regarding specific interventions such as mechanical ventilation or CPR can then be made. Having the family see the patient via video chat can be helpful in explaining the seriousness of illness or prognosis. Establishing regular video visits for family can help ease distress for both loved ones and the patient.

Case Study 7: Telephone Death Notification [36–38]

Your patient is in cardiac arrest in the ED. Resuscitation was unsuccessful. The patient is a nursing home patient with past medical history of colon cancer and dementia; it is unknown if the family has been informed. You call the next of kin.

Identify When calling next of kin, first introduce yourself, and inquire if the nursing home has already called. If this

patient was admitted from home, ask if they knew the patient was in the hospital. Again, this gives you a marker for where to begin the conversation.

Clarify Sometimes the information can be incorrect, or the person listed defers to another family member. Therefore, first clarify the relationship between the patient and the person on the phone. State that the patient is seriously ill and ask if they are able to come to the hospital. Ask how and when they are arriving, and encourage them to have another person accompany them, if possible.

****If at this point the family member is unable to get to the ED and the news must be delivered over the phone, take the following steps. ****

Ability If information must be delivered over the phone, ask if they are somewhere where they can sit down and talk. If not (e.g., they are driving or in a public space) make a definitive time to call them back, sooner rather than later.

Headliner State you have difficult news. Briefly, in one or two sentences, summarize what happened before the patient reached the ED today. Gather this information from the patient's history.

Deliver Briefly state the patient's ED course and without using jargon, state the outcome. Using the word "dead" or "died" is essential. "We attempted to restart her/his heart and help her/his breathing with breathing tubes and medicines, but despite our best efforts your [name relation to patient] died. I'm sorry."

Pause Offer condolences. Resist the urge to fill the silence with medical information.

Support Once you feel they are able to discuss more, ask for any questions, and allow for their emotional reaction as a natural response to the news.

Case Study 8: Death of a Child [39, 40]

A 6-year-old male child was admitted in cardiac arrest. The child was found by her mother in bed unresponsive; CPR was started. The response time to the hospital was 10 min. In ED, CPR continued; the patient was intubated and ACLS performed, but the patient remained asystolic. His mother is able to tell you that he was born prematurely with a seizure disorder. At age 3 he developed asthma and then leukemia. He has had multiple admissions for his complex medical problems. His leukemia was in remission after several treat-

ments. Resuscitation efforts were continued without success for 10 min; you pronounced him dead at 9:04 a.m.

Before You Go in the Room The death of a child is tragic and traumatic for all involved in his care. 22% of pediatric deaths occur in EDs [39]; therefore it is likely that we will all experience this multiple times during our careers.

Establish a private place to inform loved ones, even in situations where the family witnessed the resuscitation. Have a staff member, not yourself, accompany the loved ones to a family room. It's important to avoid an emotional confrontation in the hallway or in front of other patients or children. This staff member can also identify the head of the family, limit who should be in the room, and describe the family's emotional state. Go in with a small team. Pediatric deaths affect everyone working in the ED, and having a few of your colleagues with you for the family discussion is helpful for everyone.

In the Room Introduce yourself and identify the staff members with you. Then, carefully and slowly, tell them you have tragic news. Be succinct, tell them despite your best efforts their little boy has died, and then pause.

Expect heightened emotional reactions [41, 42]. Most adults have the inherent need to protect children, so with the death of a child, their loss can sometimes manifest itself as a sense of shame. Rarely, this is acted out in anger. Always try to have colleagues present and a safe exit strategy from the room.

Other loved ones may have many questions surrounding the death, especially if it was unexpected. Listen carefully: do they need more information, or are they experiencing an emotional reaction? It is vital to continue to listen and convey your sympathy. Keep answers as short as possible. Assure families that they are not to blame for their loss. Expect the unexpected reaction. Keep the door open or unblocked in case of a violent or flight reaction from loved ones [41]. Realize anger is not directed at you personally. Most families will question their last interactions with their child and what they could have done differently. Bring a chaplain if available.

After Discussion After delivering the news, hold a debriefing session with your ED team. Team members need to understand their emotions, as well as help one another validate each other's reactions.

Some Additional Tips

Communicating with Children

We should use developmentally appropriate language to answer a child's questions about the condition of a loved one and the circumstances surrounding their absence from the home. It is crucial to tell children the truth and not craft a less alarming reason for their absence or illness/death, such as "Grandpa went to sleep" or "has a cold." This may result in the child having difficulty trusting adults in the future or becoming scared to go to sleep or alarmed when a loved one becomes ill [43].

Child life specialists and social workers can provide guidance under these circumstances and provide literature and education to parents dealing with difficult news. Palliative care teams often have staff specializing in pediatric needs.

Explore resources that are available for parents with the assistance of the ED interdisciplinary team. The "Parenting at a Challenging Time" or PACT program offers an online toolkit to support children of parents with a serious illness that includes communication tips (https://www.massgeneral.org/assets/MGH/pdf/cancer-center/pact-toolkit.pdf) [44].

It also may be helpful for parents to engage the counselor at their school and/or their pediatrician to provide ongoing counseling and support.

Communication at Sign Out

Important considerations at sign out are in addition to preexisting checklists and required safety checks, such as name, location, and allergies, to name a few. We should include goals of care, capacity, next of kin, and plan of care (Table 62.5) [45, 46].

Takeaways: Keeping the Connection

In the ED, dealing with the "sickest of the sick" on a daily basis can be exhausting, especially to clinicians with limited training in delivering difficult news to patients/families, end of life care, or psychosocial care. For each low acuity patient that arrives in the ED, there might be two or three nursing home residents with "failure to thrive" or sepsis. Depersonalization, cynicism, and exhaustion can set in with studies showing emergency medicine clinicians at highest risk [47]. The challenge is to connect with our patients, maintain empathy, and be professional. To create a supportive environment for clinicians, there are four areas to consider: connection to patient's families through communication

 Table 62.5
 Tips for sign out/hand off [45, 46]

Any goals of care discussions to date and treatment plan

Time frame of a time limited trial of a particular treatment

Review of the surrogate medical decision-maker(s)/healthcare proxy and their corresponding contact information

Any contact and recommendations of outpatient providers, including the medical oncologist

Review of resources that have assisted with the care to date and their next steps, including palliative care consult

Most recent communication with the patient and/or family and outline of future updates or further discussion

skills; reflection of both self and department; and engagement with peers [47–51].

Reflection and Engagement [47, 48]

The ED works as a team; communicating with colleagues and other providers is essential. A natural response after a difficult case is to immerse ourselves in an endless pile of paperwork without reflection. To help counter these clinical work patterns, providers can adapt brief steps that can be incorporated into practice:

- Debriefing after every resuscitation or difficult case.
- Moment of silence after every death. No exceptions.
- *Leaving the department for a break on a clinical shift*. No matter how short.
- *Death Rounds*. Creating a ritual during faculty meetings or resident conferences around department deaths can also be useful for staff.

The Words We Use: Cultivating Patient-Centered Care

Our job in the ED is to determine what is best for our patients, medically and emotionally. Providing education on hospice and palliative medicine, which includes communication skills, to all ED clinicians including nurses, residents, and social workers, can focus care based on the patient's wishes. Find supportive palliative care staff who can champion patient-centered care within your department. Many education models exist. To launch this process clinically, you can begin by:

- *Using words* such as quality of life, patient's wishes, and comfort-focused care
- Avoiding words such as withholding, withdrawing, and phrases such as "nothing we can do"
- Expressing your thoughts and feelings when meeting with families and patients such as "I'm worried about..." or "I wish..." [23]

Actions such as these establish an environment of being conscientious and willing to engage. Integration of these communication skills and principles is realizing there is no magical finish line where all clinicians will gain resilience or cultural change. Rather, these practices require constant practice in order to maintain a "muscle memory" of patientfocused care.

References

- Baile WF, Aaron J. Patient-physician communication in oncology: past, present, and future. Curr Opin Oncol. 2005;17(4):331–5.
- Tulsky JA. Beyond advance directives: importance of communication skills at the end of life. JAMA. 2005;294(3):359–65.

- Zolnierek KB, Dimatteo MR. Physician communication and patient adherence to treatment: a meta-analysis. Med Care. 2009;47(8):826–34.
- Stewart MA. Effective physician-patient communication and health outcomes: a review. CMAJ. 1995;15(9):1423–33.
- Travaline JM, Ruchinskas R, D'Alonzo GE Jr. Patientphysician communication: why and how. J Am Osteopath Assoc. 2005;105(1):13–8.
- Cassel EJ. The nature of suffering and the goals of medicine. N Engl J Med. 1982;306(11):639–45.
- Cassell EJ. The nature of suffering and the goals of medicine. New York: Oxford University Press; 1991. p. 30–65.
- 8. Cassell EJ. Recognizing suffering. Hast Cent Rep. 1991;21(3):24-31.
- 9. Platt FW, Keller VF. Empathic communication: a teachable and learnable skill. J Gen Intern Med. 1994;9(4):222–6.
- Cassell EJ. Doctoring: the nature of primary care medicine. New York: Oxford University Press; 1997. p. 121.
- Trzeciak S, Mazzarelli A. Compassionomics: the revolutionary scientific evidence that caring makes a difference. Pensacola: Studer Group; 2019.
- Suchman AL, Markakis K, Beckman HB, Frankel R. A model of empathic communication in the medical interview. JAMA. 1997;277(8):678–82.
- Butow PN, Kazemi JN, Beeney LJ, Griffin AM, Dunn SM, Tattersall MH. When the diagnosis is cancer: patient communication experiences and preferences. Cancer. 1996;77(12):2630–7.
- Wang DH. Beyond code status: palliative care begins in the emergency department. Ann Emerg Med. 2017;69(4):437–43. https://doi.org/10.1016/j.annemergmed.2016.10.027.
- DeSandre P, Quest TE, editors. Palliative aspects of emergency care. Oxford American Palliative Care Library. Portnoy RK, series editor. New York: Oxford University Press; 2013.
- George N, Barrett N, McPeake L, Goett R, Anderson K, Baird J. Content validation of a novel screening tool to identify emergency department patients with significant palliative care needs. Acad Emerg Med. 2015;22(7):823–37. https://doi.org/10.1111/acem.12710.
- 17. Lamba S, DeSandre PL, Todd KH, Bryant EN, Chan GK, Grudzen CR, et al. Improving palliative Care in Emergency Medicine Board. Integration of palliative care into emergency medicine: the improving palliative Care in Emergency Medicine (IPAL-EM) collaboration. J Emerg Med. 2014;46(2):264–70.
- Quest T, Herr S, Lamba S. Weissman D; IPAL-EM advisory board. Demonstrations of clinical initiatives to improve palliative care in the emergency department: a report from the IPAL-EM initiative. Ann Emerg Med. 2013;61(6):661–7.
- Lunney JR, Lynn J, Hogan C. Profiles of older Medicare decedents. J Am Geriatr Soc. 2002;50(6):1108–12. https://doi. org/10.1046/j.1532-5415.2002.50268.x.
- Etchells E, Darzins P, Silberfeld M, Singer PA, McKenny J, Naglie G, et al. Assessment of patient capacity to consent to treatment. J Gen Intern Med. 1999;14(1):27–34.
- Sessums LL, Zembrzuska H, Jackson JL. Does this patient have medical decision-making capacity? JAMA. 2011;306(4):420–7.
- 22. Boehmer KR, Gionfriddo MR, Rodriguez-Gutierrez R, Dabrh AM, Leppin AL, Hargraves I, et al. Patient capacity and constraints in the experience of chronic disease: a qualitative systematic review and thematic synthesis. BMC Fam Pract. 2016;17(1):127.
- VitalTalk. Serious News: Breaking bad news using the GUIDE tool. 2019 https://www.vitaltalk.org/guides/serious-news. Accessed 18 Sept 2020.
- Aberger K, Wang D, editors. Palliative skills for frontline clinicians: case vignettes in everyday hospital medicine. Cham: Springer Nature Switzerland; 2020. p. 19, 21, 203. https://doi. org/10.1007/978-3-030-44414-3.

- 25. Cooper RS, et al. AMEN in challenging conversations: bridging the gaps between faith, hope, and medicine. J Oncol Pract. 2014;10(4):e191–5.
- 26. Mori M, Morita T, Igarashi N, Shima Y, Miyashita M. Communication about the impending death of patients with cancer to the family: a nationwide survey. BMJ Support Palliat Care. 2018;8(2):221–8.
- 27. Shinjo T, Morita T, Hirai K, Miyashita M, Sato K, Tsuneto S, Shima Y. Care for imminently dying cancer patients: family members' experiences and recommendations. J Clin Oncol. 2010;28(1):142–8.
- Lamba S, Bryczkowski S, Tyrie L, Weissman DE, Mosenthal AC. Death disclosure and delivery of difficult news in trauma #305. J Palliat Med. 2016;19(5):566–7. (Palliative Care Network of Wisconsin: Fast Facts and Concepts. https://www.mypcnow.org).
- 29. Society for Academic Emergency Medicine Ethic Committee. Death and bad news notification. Ethics in the trenches: a hero's guide. [Internet] https://www.saem.org/docs/default-source/ education/death-and-bad-news-notification.pdf?sfvrsn=0.
- Weissman DE, Quill TE, Arnold RM. Preparing for the family meeting #222. J Palliat Med. 2010;13(2):203–4. (Palliative Care Network of Wisconsin: Fast Facts and Concepts. https://www. mypcnow.org).
- Chaitin E, Rosielle DA. Responding to requests for non-disclosure of medical information #219. Palliative care network of Wisconsin: fast facts and concepts. https://www.mypcnow.org
- Christakis NA, Lamont EB. Extent and determinants of error in physicians' prognoses in terminally ill patients: prospective cohort study. West J Med. 2000;172(5):310–3.
- 33. Smith-Uffen MES, Johnson SB, Martin AJ, Tattersall MHN, Stockler MR, Bell ML, et al. Estimating survival in advanced cancer: a comparison of estimates made by oncologists and patients. Support Care Cancer. 2020;28(7):3399–407.
- Lamont EB, Christakis NA. Prognostic disclosure to patients with cancer near the end of life. Ann Intern Med. 2001;134(12):1096–105.
- VitalTalk. COVID Ready Communication Playbook. 2019. https://www.vitaltalk.org/guides/covid-19-communication-skills/. Accessed 18 Sept 2020.
- Hobgood C, Harward D, Newton K, Davis W. The educational intervention "GRIEV_ING" improves the death notification skills of residents. Acad Emerg Med. 2005;12(4):296–301.
- Osias RR, Pomerantz DH, Brensilver JM. #76 telephone notification of death – part 1. Palliative care network of Wisconsin: fast facts and concepts. www.mypcnow.org
- Osias RR, Pomerantz DH, Brensilver JM. #76 telephone notification of death – part 2. Palliative care network of Wisconsin: fast facts and concepts. www.mypcnow.org

- 39. Committee on approaching death: addressing key end of life issues; institute of medicine. Dying in America: improving quality and honoring individual preferences near the end of life. Washington, DC: National Academies; 2015. Appendix F, Pediatric End-of-Life and Palliative Care: Epidemiology and Health Service Use. https:// www.ncbi.nlm.nih.gov/books/NBK285690/
- Adamowski K, Dickinson G, Weitzman B, Roessler C, Carter-Snell C. Sudden unexpected death in the emergency department: caring for the survivors. CMAJ. 1993;149(10):1445–51.
- VitalTalk. Conflicts. Defusing conflicts: moving the conversation forward. 2019. https://www.vitaltalk.org/guides/conflicts/. Accessed 18 Sept 2020.
- 42. Jelinek GA, Marck CH, Weiland TJ, Philip J, Boughey M, Weil J, Lane H. Caught in the middle: tensions around the emergency department care of people with advanced cancer. Emerg Med Australas. 2013;25(2):154–60.
- 43. American Academy of Child and Adolescent Psychiatry. Parents with cancer: tips for talking to kids. https://www.aacap.org/ AACAP/Families_and_Youth/Facts_for_Families/FFF-Guide/ When-a-Parent-Has-Cancer-Tips-for-Talking-to-Kids-117.aspx. No. 117. Updated May 2019. Last Accessed 18 Sept 2020
- Marjorie E. Korff PACT Program. Parenting at a challenging time. Massachusetts general hospital cancer center. https://www. mghpact.org/. Accessed 27 Jun 2020.
- 45. Cheung DS, Kelly JJ, Beach C, Berkeley RP, Bitterman RA, Broida RI, et al. Section of quality improvement and patient safety, American College of Emergency Physicians. Improving handoffs in the emergency department. Ann Emerg Med. 2010;55(2):171–80.
- Meier DE, Beresford L. Palliative care's challenge: facilitating transitions of care. J Palliat Med. 2008;11(3):416–21.
- 47. Lu DW, Dresden S, McCloskey C, Branzetti J, Gisondi MA. Impact of burnout on self-reported patient care among emergency physicians. West J Emerg Med. 2015;16(7):996–1001.
- 48. Mull CC, Bowman WR. A call to restore your calling: self-care of the emergency physician in the face of life-changing stress: part 5 of 6: physician burnout. Pediatr Emerg Care. 2020;36(1):e25–9.
- 49. Philip J, Remedios C, Breen S, Weiland T, Willenberg L, Boughey M, et al. The experiences of patients with advanced cancer and caregivers presenting to emergency departments: a qualitative study. Palliat Med. 2018;32(2):439–46.
- 50. Committee on approaching death: addressing key end of life issues; institute of medicine. Dying in America: improving quality and honoring individual preferences near the end of life. Washington, DC: National Academies Press; 2015.
- Lee SJ, Back AL, Block SD, Stewart SK. Enhancing physicianpatient communication. Hematology Am Soc Hematol Educ Program. 2002;2002:464–83.

Paul L. DeSandre and Gretchen E. Bell

Introduction

Research addressing the emergency department (ED) care of the seriously ill is growing. At the time of this writing, a simple PubMed (NLM) search for articles combining the "Title" terms "Palliative" and "Emergency" shows a marked increase since 2008 (Fig. 63.1). In order to better understand relevant historical studies and identify future directions for palliative care (PC) research in the ED, we expanded our search and identified three articles addressing research opportunities, priorities, and cancer-specific concerns. Those articles span a range of 10 years with common themes, providing the structure for this review of relevant research in the field.

The first of those three articles appeared in 2006 at the beginning of the increase in related publications. This article provides a list of research opportunities for emergency clinicians derived from consensus statements on end-of-life care from the 2004 National Institutes of Health state-of-the-science conference and the National Institute of Nursing Research 2006 Areas of Research Opportunity [1]. The next article, published in 2011, defined a set of research priorities for PC in the ED. These priorities for research were the product of an expert consensus workgroup conference titled "Improving the Quality and Efficiency of Emergency Care Across the Continuum: A Systems Approach," sponsored by the Agency for Healthcare Research and Quality (AHRQ)

Department of Emergency Medicine, Emory University School of Medicine, Atlanta, GA, USA e-mail: pdesandre@emory.edu

G. E. Bell Department of Family and Preventive Medicine, Emory University School of Medicine,

Atlanta, GA, USA

and the American College of Emergency Physicians (ACEP). From this workgroup, four key research questions emerged: Who are the patients requiring PC in the ED? What is the role of the emergency clinician for these patients? What is the impact of PC integration in health care utilization? What are the educational priorities?

The six categories of research priorities proposed to address these questions were descriptive, attitudinal, screening, outcomes, resource allocation, and education of clinicians [2]. The third article, from 2016, offered a specific research agenda for the emergency care of patients with cancer. In this agenda, important PC research needs focused on addressing health care use and patient management [3]. For this chapter, in order to include a broader perspective beyond cancer patients in the ED, we use the six categories of research priorities from the second article as a framework. We provide further categorization using evidence-based medicine terminology to describe research methods as described by Greenhalgh [4]. With this combined structure, our goal is to present essential information on the growing body of research devoted to improving the care of patients with serious illness in the ED (Table 63.1).

Descriptive Studies

Descriptive studies of PC in the ED provide information on existing characteristics of patients, circumstances of their care, and unmet needs. Studies of this type began around 2002 with substantial increases in recent years. The vast majority of these studies are of cohort design. Descriptive studies should provide an accurate depiction of patients with serious illness and their prehospital and ED experience. In reviewing the related literature, four categories emerge: (1) general PC; (2) cancer; (3) pediatric PC; and (4) coordination and access to care.



Palliative Care Research

P. L. DeSandre (\boxtimes)



Number of Annual PubMed Citations including "Palliative" and "Emergency" in Title

Fig. 63.1 Search of PubMed (National Library Medicine) for articles combining the "Title" terms "Palliative" and "Emergency" showing a marked increase since 2008

General Palliative Care

While the range of topics is broad and the methods variable, general descriptive PC research has focused on defining patient characteristics and the provision of PC in the ED. Two systematic reviews describe available findings, one in 2015 [5] and another in 2019 [6]. The first addresses the prevalence of advance directives in the ED. The second describes the impact of PC services on ED use. In the first article, Oulton et al. found an overall low rate of advance directives completion with wide variation. Certain variables were associated with a higher likelihood of completion of advance directives. These included patients who were older, were in poor health, belonged to a specific religion, were widowed, had children, were a resident of an institution, or had a primary care provider. These characteristics suggest awareness and understanding of the consequences of illness, which may have encouraged the completion of advance directives at some point prior to needing ED evaluation. The second systematic review by Bone et al. looked at the impact of PC services on ED use and found that the provision of PC or hospice services was associated with lower ED use in the last

year of life for patients \geq 65 years of age. The majority of included studies were from the United States with almost no information from low- to middle-income countries.

The remaining publications were of either cohort or crosssectional design. These articles address the range of models of provision of care and the substantial PC needs in the ED. Quest, et al. describe four types of ED PC integration based on 11 successful programs: traditional consultation, basic integration, advanced integration, and ED-focused advanced integration [7]. A 2010 single-site cross-sectional survey of 50 functionally impaired adults \geq 65 years old in the ED found the majority of patients exceeded severity of needs assessments for physical symptoms, finances, mental health, and access to care [8]. In 2012, patients dying in EDs in France and Belgium were reported as having a lack of palliative interventions despite 80% having died in the context of withholding or withdrawing of life-sustaining interventions [9]. From the same year, a US study from a large urban ED reported that 84% of the patients seen by PC consultants died while in the ED. Common patient characteristics for consultation included young age, sudden critical illness or traumatic event, or were imminently dying [10]. Yet another

 Table 63.1
 Palliative care research in the emergency department [2, 4]

Research priority ^a	Year	First author	Objectives	Type of Study ^b
Descriptive				
General palliative	2015	Oulton [5]	Determine the prevalence of ADs among ED patients, with a focus on	Systematic review
care			older adults, as well as factors associated with rates of AD completion	
	2019	Bone [6]	Review factors associated with ED attendance during the last year of life in older adults	Systematic review
	2013	Quest [7]	Describe clinical examples of ED and PC integration, including	Cross-sectional
	2010	Crudzon [9]	ED-provided subspecially PC practice	Cabort
	2010	Von Tricht [0]	Describe properties of duing national who received PC in the ED	Cross sectional
	2012	Vali Illelit [9]	Describe proportion of dying patients who received FC in the ED	Cobort
	2012		why ED-PC consults are currently requested	
	2017	Blackwell [11]	Critique the feasibility of experience-based co-design of a methodology for quality improvement interventions to improve PC experiences in the ED	Qualitative: experience-based co-design
	2015	George [12]	Develop a simple content valid screening tool for use by EPs to identify patients with significant PC needs and establish content validity	Qualitative: modified Delphi technique with expert panel
	2019	Köstenberger [13]	Determine the prevalence of patients presenting with palliative symptoms in an Austrian ED	Cohort
Cancer	2020	Chen [14]	Describe characteristics of cancer patients who utilize emergency medical services	Cohort
	2010	Barbera [15]	Describe the most common reasons for visits made to the ED during the final 6 months and the final 2 weeks of life in patients who die of cancer	Cohort
	2011	Mayer [16]	Examine why patients with cancer present to the ED	Cohort
	2015	Henson [17]	Explore factors associated with ED attendance by cancer patients in their last month of life	Systematic review and meta-analysis
	2016	Elsayem [18]	Examine the presenting symptoms associated with risk of ICU admission and hospital death in patients with cancer admitted through an ED	Cohort
	2017	Rivera [19]	Estimate the proportion of ED visits made by cancer patients, understand their clinical presentation and examine factors related to inpatient admission. National dataset	Cohort
	2018	Henson [20]	Identify socioeconomic and clinical factors associated with ED attendance at EOL and the relationship with prior ED use among cancer decedents	Cohort
	2019	Caterino [21]	Understand common characteristics of patients with cancer seeking emergency care and identify opportunities for care optimization	Cohort
	2019	Verhoef [22]	Compare EOL trajectories of hematologic malignancy and solid tumor patients who died <3 months after ED visit	Cohort
Pediatric palliative care	2018	Gaucher [23]	Describe the characteristics of pediatric PC patients with pediatric ED visits	Cohort
	2018	Rocha [24]	Identify factors associated with emergency visits and hospitalizations in patients with congenital syndrome of Zika virus. Identify the clinical interventions performed from the PC perspective	Cross-sectional
Coordination and access to care	2002	Ausband [25]	Determine the prevalence of PC protocols among EMS agencies in the United States and what proportion of the population is covered by those protocols	Cross-sectional
	2014	Carron [26]	Describe situations where prehospital physicians managed PC situations	Case report
	2013	Wallace [27]	Identify patients receiving specialist PC services who attend the ED and determine if these presentations are potentially avoidable	Cohort
	2013	Hjermstad [28]	Register the reasons patients with advanced cancer attend the ED, the interventions performed during their hospital stay, and symptom intensity upon admission and discharge. Assess patient attitudes toward ED attendance	Qualitative: mixed cohort and structured interviews
	2018	Kao [29]	Investigate the factors for ED use during out-of-hours periods of palliative home care services among advanced cancer patients	Cohort
	2018	Lipinski [30]	Evaluate the use of PC services in patients with heart failure who present to the ED	Cohort

(continued)

Table 63.1	(continued)
------------	-------------

(continue	ucu)			
Research priority ^a	Year	First author	Objectives	Type of Study ^b
	2019	Green [31]	Examine the decision-making process that leads to patients with palliative care needs to seek emergency care	Qualitative: narrative interview
	2014	Shin [32]	Compare the symptom burden and survival rate of patients admitted to an acute palliative care unit from an ED to those transferred from an inpatient unit	Cohort
	2017	Spilsbury [33]	Determine how the association of community-based PC with \downarrow ED visits in the last year of life varied by patient factors	Cohort
Attitudinal				
Provision of palliative care in the ED	2009	Wiese [34]	Explore the attitudes and beliefs of prehospital EPs about their experiences in dealing with PC patients in out of hospital emergency situations	Cohort
	2012	Gruzden [35]	Explore attitudes and beliefs among attending EP's provision of PC services in ED	Qualitative: semi-structured focus groups
	2013	Lamba [36]	Elicit the perceived barriers by EP to providing PC in the ED	Cross-sectional
	2015	Weil [37]	Explore the understanding of PC by healthcare professionals caring for patients with advanced cancer attending the ED	Qualitative: focus groups and semi- structured interviews
	2019	Côté [38]	Propose implementation strategies for pediatric palliative care in the ED	Qualitative: semi-structured focus groups
	2009	Smith [39]	Explore the attitudes, experience, and beliefs of EPs about PC in the ED	Qualitative: focus groups
	2015	Russ [40]	Investigate staff experience and attitude toward PC provision in a public metropolitan ED	Cross-sectional
	2018	Wright [41]	Identify EP's priorities for improvement in ED-based PC for older adults	Qualitative: semi-structured interviews
	2013	Grudzen [42]	Identify hospital level factors from the administrative perspective that affect the availability and delivery of PC services in the ED	Qualitative: semi-structured interviews
	2015	Rivera [43]	Explore the perceptions and barriers encountered by practicing EPs in providing PC in Puerto Rican EDs	Cross-sectional
	2019	Argintaru [44]	Describe perceived barriers and facilitators to GOC discussions by EM physicians/residents	Cross-sectional
	2010	Smith [45]	Understand the experiences of acutely symptomatic, terminally ill patients seen in the ED	Qualitative: semi-structured interviews
	2018	Cooper [46]	Raise awareness of the experience of PC patients and families/ caregivers together with the insights of clinicians caring for them increasing understanding of why implementation of good practice remains difficult	Systematic review
	2019	Di Leo [47]	Explore issues with delivering ED-PC from the perspective of both providers and patients	Qualitative: focus groups and semi- structured interviews
End-of-life care in the ED	2017	Fernandez-Sola [48]	Define the attributes of dignity in EOL care in the ED, based on the opinions of physicians/nurses	Qualitative: focus groups and semi- structured interviews
	2019	Giles [49]	Examine the experiences, perceptions, and needs of nurses providing EOL care in the ED setting in relation to sudden and unexpected deaths	Qualitative: descriptive survey
	2019	Alqahtani [50]	Explore challenges from a staff perspective of safe appropriate high-quality EOL care for people diagnosed with non-malignant diseases who present to the ED	Systematic review
	2020	Mughal [51]	Describe, understand, and explain the perceptions of ED nurses on provision of EOL care	Systematic review
Educational needs	2011	Meo [52]	Characterize the level of formal training and perceived educational needs in PC of EM residents	Cross-sectional survey
	2012	Lamba [53]	Characterize EP's perceived educational and formal training needs for PC-related skills	Cross-sectional survey

Table 63.1 (continued)

Research priority ^a	Year	First author	Objectives	Type of Study ^b
Screening				a
General	2020	Kistler [54]	Identify existing triggers for PC consult, how they compare between settings, and common criteria used to identify those in need of PC consult	Systematic review
	2008	Mahony [55]	Identify chronically ill older patients with PC, homecare or hospice needs, and \uparrow linkage with these services	Cohort
	2016	George [56]	Evaluate the methods, tools, and outcomes of PC screening and referral projects in the ED	Systematic review
	2001	Mion [57]	Improve case finding of at-risk older adults and provide comprehensive assessment in the ED setting with formal linkage to community agencies	Cohort
	2011	Glajchen [58]	Improve PC referral among frail, elderly patients in the ED	Cohort
SPEED	2011	Richards [59]	Develop and validate a novel palliative medicine needs assessment tool for patients with cancer in the ED	Cohort
	2019	Reuter [60]	Assess the feasibility and reach of a novel care pathway utilizing electronic health record embedded screening and automatic triggered consultation to deliver ED-based palliative care services for patients with active cancer	Cross-sectional
P-CaRES	2015	George [12]	Develop a simple content valid screening tool for use by EPs to identify patients with significant PC needs and establish content validity	Qualitative: modified Delphi technique with expert panel
	2016	Bowman [61]	Assess acceptability and reliability of a brief novel content – validated screening tool for unmet palliative care needs – the Palliative Care and Rapid Emergency Screening (P-CaRES) Project, a brief novel content among EPs of different roles and experience levels	Cross-sectional
	2017	Ouchi [62]	Assess the performance and determine the acceptability of a content- validated palliative care screening tool	Cohort
	2020	Tan [63]	Optimize the use of electronic health records by creating a clinical decision support (CDS) tool to identify the high-risk patients most likely to benefit from primary PC and provide point-of-care clinical recommendations	Qualitative: focus groups and survey
Surprise question	2018	Ouchi [64]	Evaluate the prognostic value of the suprise question (SQ) in elderly ED (>65) patients for predicting 12-month mortality	Cohort
	2019	Verhoef [65]	Evaluate the prognostic value of the surprise question (SQ) in patients with advanced cancer visiting the ED and study the yield of adding other predictors for approaching death	Cohort
	2019	Aaronson [66]	Evaluate the prognostic value of the SQ in symptomatic heart failure patients presenting to the ED	Cohort
	2019	Ouchi [67]	Determine the association of the surprise question with actual 1-month mortality among undifferentiated older adults admitted from the ED	Cohort
	2019	Haydar [<mark>68</mark>]	Determine the utility of the modified SQ (30 days versus 1 year) to predict in-hospital mortality and resource utilization for hospitalized ED patients	Cohort
Other ED screening projects	2014	Ouchi [69]	Observe prevalence of ED-initiated palliative care consultation for advanced dementia following quality improvement intervention	Cohort
	2015	Kistler [70]	Evaluate the impact of ED-initiated palliative care referral on the proportion and timing of consultation	RCT
Outcomes				
General	2019	Grudzen [71]	Measure the impact of primary PC education for ED providers on disposition, healthcare utilization, and survival times	RCT
	2020	Yash [72]	Measure the current quality of EOL care in ED to identify gaps, formulate improvements, and implement the improved EOL care protocol	Cohort
	2014	Lamba [73]	Highlight needs and opportunities for better integration of EM and PC	Expert opinion
	2016	da Silva Soares [74]	Examine the effectiveness of ED-based PC interventions on hospital admissions, LOS, symptom severity, QOL, use of other healthcare services, and PC referrals for adults with advanced disease	Systematic review
	2020	Wilson [75]	Assess the effect of ED-based PC interventions on patient- and family-reported outcomes, healthcare utilization, and survival for adult patients	Systematic review

(continued)

Table 63.1 (continued)

	licu)			
Research priority ^a	Year	First author	Objectives	Type of Study ^b
Traditional model of consultation	2016	Delgado-Guay [76]	Assess the effect of ED-PC consultation in advanced cancer patients on time to PC consult, symptom intensity, LOS, and disposition	Cohort
	2019	Mogul [77]	Investigate the resource utilization and mortality of patients age ≥ 65 from long-term care facilities who present as emergency severity index level 1 (most urgent)	Cohort
	2019	El Majzoub [78]	Examine the association between timing of palliative care consult and in-hospital mortality in admitted cancer patients	Cohort
Basic integration	2016	Grudzen [79]	Assess the effects of ED-initiated PC consultation in patients with advanced cancer on QOL, depression, healthcare utilization, and survival	RCT
	2016	Highet [80]	Evaluate the feasibility of a focus intervention in the ED to identify hospice-eligible patients and facilitate disposition	Cohort
	2019	Koh [81]	Study the feasibility of a three-way model of care between the ED, inpatient PC team, and hospice services designed to increase early access to palliative care	Cohort
Advanced integration	2019	Tiernan [82]	Evaluate if a multifaceted intervention composed of flagging known patients, screening tool, proactive palliative medicine engagement in the ED, and educational programming impacts palliative care referral rates or healthcare utilization metrics	Cohort
	2017	Weng [83]	Improve the quality of care for ED patients at EOL using PC champions, education, and close collaboration with a hospice team	Cohort
ED-focused advanced integration	2013	Rosenberg [84]	Describe an approach in expanding the role of ED staff into palliative and EOL care	Case report
	2015	Grudzen [85]	Transform geriatric emergency care by applying PC principles via a multi-step process improvement	Cohort
	2018	Liberman [86]	Investigate the effectiveness of an Advanced Illness Management program in the ED on improving outcomes	Cohort
End-of-life care	2008	Sedillot [87]	Describe a five-step protocol for withholding and withdrawing of life support in an ED for terminally ill patients	Cohort
	2013	Shlamovitz [88]	Discuss palliative sedation in the ED and the use of ketamine	Case report
	2019	Wang [89]	Discuss best practice through a systematic approach to comfort care transitions in the dying ED patient	Overview: non- systematic review
	2019	Economos [90]	Assess the appropriateness of EOL care provided to actively dying patients in the ED	Cohort
	2019	Chor [91]	Describe the epidemiological characteristics, symptom burden, and management of patients using a protocolized management care bundle	Cohort
	2020	Ruangsomboon [92]	Assess whether high-flow nasal cannula is superior to conventional oxygen in relieving dyspnea in PC patients in the ED with do not intubate (DNI) status	RCT
Communication	2018	Mills [93]	Evaluate the utility to doctors of a form specifically designed to guide and document GOC discussions at point of care	Cohort
	2019	Hanning [94]	Determine which patients benefit, requirements, content, documentation, and harms and benefits for ED GOC discussions	Systematic review
	2019	Ouchi [95]	Describe development of a brief negotiated interview ED intervention to increase engagement in serious illness conversation by seriously ill older adults and describe its acceptability	Cohort
	2018	Leiter [96]	Assess whether trained EPs can administer a structured brief negotiated interview intervention to facilitate advance care planning conversations with high fidelity	Cohort
Resource allocation				
Community-based	2020	Taylor [97]	Evaluate PC initiatives which may avoid or shorten hospital stay at the end of life and analyzed their success in terms of reducing bed days	Overview: non- systematic review
	2008	Wiese [98]	Demonstrate the influence of PC teams on emergency calls by cancer patients or their relatives during the last 6 months of life	Cohort
	2016	Obermeyer [99]	Compare patterns of ED use and inpatient admission rates for elderly adults with cancer with a poor prognosis who enrolled in hospice to those of similar individuals who did not	Case-control
Table 63.1 (continued)

Research priority ^a	Year	First author	Objectives	Type of Study ^b
	2017	Sutradhar [100]	Examine the association between palliative versus standard homecare nursing and the rate of high-acuity and low-acuity ED visits among cancer decedents during their last 6 months of life	Cohort
	2018	Hirvonen [101]	Explore the effect of regional efforts to develop EOL care pathways and the availability of outpatient palliative clinic on the number, quality, and outcome of ED visits toward the EOL for patients with cancer	Cohort
	2013	Lamba [102]	 Review four case scenarios that relate to palliative care and may be commonly encountered in the out-of-hospital setting Provide a road map by suggesting four things to do to start an EMS-palliative care initiative in order to optimize out-of-hospital care of the seriously ill and increase preparedness of EMS providers in these difficult situations 	Expert opinion
	2009	Wiese [103]	Explore the attitudes and beliefs of prehospital emergency physicians about their experiences in dealing with PC patients in out of hospital emergency situations	Cohort
	2017	Montgomery [104]	Develop an innovative strategy to provide collaborative care in the home to alleviate symptoms and avoid transport	Cohort
	2019	Kamphausen [105]	Investigate challenges faced by EPs who provide prehospital emergency care to patients with advanced incurable diseases and family caregivers in their familiar home environment	Qualitative: semi-structured interviews
	2019	Patterson [106]	Explore the extent to which access to, and quality of, patient information affects the care paramedics provided to patients nearing end of life and their views on a shared electronic record as a means of accessing up-to-date patient information	Qualitative: semi-structured interviews
	2017	McQuown [107]	Evaluate which patients presenting to the ED had their DNR status recognized by the physician and DNR orders that were made during their hospital stay	Cohort
	2016	Lakin [108]	Assess ED physicians' experiences with advance care planning electronic medical record documentation and their documentation needs	Cross-sectional
	2020	Vranas [109]	Evaluate how POLST form completion, treatment limitations, or both influence intensity of treatment among patients who present to the ED	Cohort
Within the ED	2015	Purdy [110]	Investigate the impact of the Marie Curie Cancer Care Delivering Choice Programme on place of death and hospital usage (ED and admissions)	Cohort
	2016	Lafond [111]	Improve the environment for EOL care of terminal ED patients	Cohort
	2020	Liberman [112]	Investigate the effectiveness of an artificial intelligence management program in the ED on key outcomes	Cohort
	2020	Chidiac [113]	Describe a system designed to optimize PC provision beyond specialist services during COVID-19	Expert opinion (letter to the editor)
	2020	Fausto [114]	Describe a multifaceted strategy to implement high-quality palliative care in the context of the COVID-19 pandemic that incorporates conventional, contingency, and crisis capacity	Expert opinion
After the ED	2020	Lee [114]	Examine the clinical characteristics and outcomes of patients who received intervention by a novel COVID-19 palliative care response team, focused on providing high-quality goals-of-care conversations in time-critical situations	Case series
	1999	Ting [117]	Study the early impact of bereavement and evaluate the effectiveness of the bereavement care given by a multidisciplinary team to close relatives of a sudden death, measured by the intensity of grief reaction (Texas Revised Inventory of Grief)	Cohort
	2019	Cooper [118]	Describe the implementation of ED Grief Support, a program developed to extend care to the bereaved through in-person, telephone, and e-mail follow-up for 1 year after the death of a loved one	Cohort
Education for clinicians				
Death disclosure	2002	Quest [119]	Design, implement, and evaluate a multidimensional, interdisciplinary, educational training module that enables residents to deliver an effective and empathic death disclosure in the emergency setting	Cohort
				(

(continued)

Table 63.1 (continued)

Research priority ^a	Year	First author	Objectives	Type of Study ^b
	2006	Quest [120]	Explore the validity and reliability of the affective competency score (ACS), compared to a global rating measure to predict overall competency to perform a death disclosure in a standardized patient exercise and investigate useful thresholds of the ACS	Cohort
	2005	Hobgood [121]	Improve resident confidence, competency, and communication skills when delivering a death notification using the GRIEV_ING mnemonic	Cohort
	2013	Hobgood [122]	Improve the confidence, competency, and communication skills of EMS personnel in death notification using the GRIEV_ING mnemonic	Cohort
EPEC-EM	2010	Gisondi [123]	Assess adaptation of a comprehensive training course in palliative and end-of-life care for an emergency medicine residency. Compare asynchronous versus synchronous learning adaptations	Case-control
	2010	Ponce [124]	Study the feasibility and utility of an educational intervention designed to improve prehospital provider comfort with family witnessed resuscitation and death notification	Cohort
	2012	DeVader [125]	Assess resident knowledge, comfort, and proficiency in key palliative/ hospice care principles and whether this can be improved via a brief education intervention	Cohort
	2018	Goldonowicz [126]	Investigate the value of a novel simulation-based palliative care educational intervention within an EM residency curriculum	Cohort
Training efforts	2016	Kraus [127]	Assess formal education on palliative care domains in emergency medicine residency programs and assess barriers/opportunities for further integration	Cross-sectional
	2019	Baylis [128]	Describe the number of Canadian postgraduate EM training programs with palliative care curricula	Cross-sectional
	2018	Shoenberger [129]	Define content areas and competencies for HPM primary-level practice in the ED	Expert opinion
	2019	Gruzden [71]	Measure the impact of primary palliative care education for ED providers on disposition, healthcare utilization, and survival times	RCT

Legend of abbreviations: AD Advanced directive, COVID-19 Coronavirus disease 2019, ED Emergency department, EOL End of life, EP Emergency physician, EPEC-EM Education in Palliative and End-of-Life Care for Emergency Medicine, ICU Intensive care unit, LOS Length of stay, PC Palliative care, P-CaRES Palliative Care and Rapid Emergency Screening, QOL Quality of life, RCT Randomized controlled trial, SPEED Screen for Palliative and End-of-life care needs in the Emergency Department, SQ Surprise question

Research priority^a

- 1. Descriptive
- 2. Attitudinal
- 3. Screening
- 4. Outcomes
- 5. Resource allocation
- 6. Education for clinicians

Type of study^b

- I. Quantitative studies
 - 1. Systematic review and meta-analysis
 - 2. RCT with definitive results
 - 3. RCT with non-definitive results
 - 4. Cohort studies
 - 5. Case-control studies
 - 6. Cross-sectional studies
- 7. Case reports
- II. Qualitative studies
- III. Other
 - 1. Overview: non-systematic review
 - 2. Expert opinion

study from 2012 compared a month of ED-initiated PC consultations 4 years apart in a large urban ED in the United States. They found that most patients were admitted through the ED, yet consultations initiated in the ED were only 3% in 2005 and 6% in 2009 [11]. A similar study from 2019 assessed ED patients with unmet PC needs in Austria, based on a previously validated screening tool [12], with only 5.5% receiving PC consultation [13].

Cancer

In the cancer population, most of the descriptive studies focus on patient characteristics and ED use. One study showed the prehospital Emergency Medical Service (EMS) use in cancer patients appears to be greater among older patients, those with higher acuity, CNS cancers, or neurological or cardiac symptoms [14]. Lung cancer is consistently found to be the most common type of presenting cancer, as well as the cancer most likely associated with in-patient death following ED use. Overall, lower socioeconomic status and symptom distress, particularly pain and dyspnea, are among the most common reasons for ED presentation [15– 20]. In a 2019 multicenter prospective cohort study from 18 EDs, Caterino et al. found that nearly half of cancer patients presenting to the ED were ≥ 65 years old, and those presenting with uncontrolled pain from outpatient settings were associated with only 8% of specialist PC involvement [21]. Comparing hematological to solid tumor malignancies, those with hematological malignancies appear to have a substantially higher likelihood of in-hospital, ICU, and ED death [22].

Pediatric

There is a relative paucity of descriptive research addressing pediatric PC in the ED. Two 2018 studies, one from Canada and the other from Brazil, demonstrate missed opportunities and the need for greater access and collaboration. The Canadian study is a retrospective review of patients followed by the pediatric PC team who presented to their ED. Of the 290 patients studied, 40% died during that hospitalization with nearly 20% dying within the first 72 hours of admission. Two-thirds had completed advance directives on arrival to the ED, but less than 40% had goals of care discussions within that hospitalization [23]. The Brazilian study assessed patients presenting to the ED with congenital Zika virus syndrome. Affected children have a limited lifespan and require frequent ED visits, hospitalizations, and aggressive procedures throughout their life. Of the 92 patients followed in their specialty clinic who presented to the ED, only two hospitalized patients received PC specialty consultation [24].

Coordination of Care and Access to Palliative Care

The remaining descriptive articles of PC in the ED come from a variety of international sources examining access and coordination of care. Two publications studied emergency medical service (EMS) issues. The first, in 2002, surveyed a cross section of EMS agencies from 121 of the 200 most populous US cities and found only 5.8% of those agencies had protocols to care for patients with PC concerns [25]. The other article from 2014 used case reports to encourage increased PC education for EMS personnel and improved collaboration with healthcare systems [26].

Five studies from several different countries demonstrate a need for better access to PC services, particularly after usual business hours and even when previously connected to outpatient PC services [27–31]. One of these studies reviewed medical records of 500 heart failure patients presenting to two EDs in Canada during an 8-month period in 2013. Among those who died within 1 year, 41% received specialist-level PC, and 76% of those patients received services only in the last 2 weeks of life [30].

Two remaining descriptive studies showed benefits of PC specialty involvement. The first compared characteristics of patients directly admitted to an inpatient acute PC unit directly from the ED (12% of 2568 patients) versus those admitted elsewhere in the hospital. They found that while the ED patients had higher symptom burden and higher medical acuity on admission to their unit, they were more likely to be discharged to the community alive [32]. The second article used a regional Australian database of deceased patients with serious illness and found a 50% reduction in ED use in the last year of life for those enrolled in the PC community-based programs versus those not enrolled in these programs [33].

Attitudinal Studies

Studies exploring attitudes and perceptions of PC in the ED have identified barriers to care and opportunities for improvement. Nearly all related studies have occurred within the past decade. The majority of these studies viewed the provision of PC in the ED through the lens of healthcare workers, patients, and caregivers. A smaller number of studies focus specifically on end-of-life care, and the remainder target emergency medicine resident training needs.

Provision of Palliative Care in the Emergency Department

An early study assessed the emergency care of patients with PC needs from the perspective of 150 prehospital emergency physicians in Germany. A large majority (89%) reported experience with patients having advanced cancer with acute PC needs, but half were uncomfortable with the associated psychosocial challenges and felt that additional training would be useful [34]. Within the ED, several studies investigated PC-related attitudes of healthcare workers. Some targeted emergency physicians specifically [35, 36], and others included interprofessional physicians involved in the care of palliative patients with ED visits [37, 38]. The remainder included a more general group of providers and healthcare workers in the ED [39-41], as well as healthcare administrators [42]. Common and informative themes emerged from this collection of studies of healthcare provider attitudes. Among emergency physicians, there is a perceived lack of understanding, lack of training, and discomfort in providing PC in the ED. The role of the emergency physician was also viewed as more fixed within a culture of stabilization rather than a focus on quality of life. Common themes for emergency physicians included recognition of the complexity of medical decision-making and medicolegal practice concerns in the ED that inhibit consideration of withholding or withdrawing life-sustaining interventions. They also communicated a concern that PC-related questions should have been addressed prior to the ED visit by a primary care provider. In contrast to these findings, a cross-sectional survey of ED residents and faculty reported a high degree of comfort with addressing goals of care and that this skill was within their scope of practice. Specialist PC clinicians in the ED are viewed overall as beneficial for their skills, time, and ability to intervene; but availability is inconsistent, and in some environments specialists are unavailable [35, 36, 43, 44]. Beyond the attitudes of emergency physicians, a broader view from ED-based healthcare providers found similar themes, such as a lack of understanding and a desire for additional training and education, but showed a greater association of PC with end-of-life care than in the physician groups. In addition to medicolegal constraints and a discordant culture of "life-saving" rather than palliation, these studies emphasize the need for greater communication and documentation to prevent undesired outcomes, such as cardiopulmonary resuscitation or intubation in a patient who would not want such interventions [39–41].

Investigations of barriers to care and opportunities for improvement widen the view further, including attitudes of interprofessional clinicians and administrative healthcare professionals, as well as patients and family caregivers. Certain barriers to providing PC in the ED were identified as similar to those seen by ED providers, such as a lack of understanding of PC, a culture of "aggressive care," and medicolegal concerns. PC in the ED was seen as desirable overall and an opportunity for improved patient and family satisfaction, more meaningful care, and decreased costs [37, 38, 42]. When including interviews with patients and family caregivers, in addition to healthcare workers' perceptions of care in the ED, common recommendations for improvement included a need for a more accommodating environment of care, better communication and coordination of care, and greater education and training among ED providers [45-47].

End-of-Life Care in the Emergency Department

Physicians and nurses provide care for the dying and deceased in the ED routinely. As would be expected, studies have demonstrated their support for preserving the dignity of patients dying in the ED, yet they also identified several challenges that threaten their ability to accomplish this goal. Among those challenges are a suboptimal physical environment of care (e.g., little space, limited privacy) and inadequate resources, communications, and coordination of care. Finally, ED provider role ambiguity and insufficient education and training are of concern. Improvement priorities included enhanced communication and symptom management skills, standardized care protocols, adequate space and resources, as well as better ED-PC integration [48–51].

Educational Needs

Two cross-sectional surveys assessed perceived educational needs of emergency medicine residents. The first, published in 2011, surveyed 159 residents at all levels of training from six emergency medicine programs in New York City. Half of those surveyed described having some PC training prior to residency. By the second year of residency, nearly all reported ED experience in managing terminal illnesses and with patients who died. PC was seen as an important competency by 71% of respondents, but only 24% had a "clear idea of the role of PC in emergency medicine." The priorities for additional training included advance directives and medicolegal issues at end of life [52].

The second study from the following year surveyed both residents and faculty from a single program. Approximately half of those surveyed felt they received minimal training in pain management, hospice care, withdrawing/withholding life-sustaining interventions, and management of the dying patient. Similar to the prior study, PC was valued as an important competence by 88% of respondents, and 90% desired more training. Specific topics identified for additional training included discussion of code status and symptom management at end of life. Bedside teaching was the preferred learning modality [53].

Screening

General

The most recent systematic review of triggered PC consults is from 2020 [54]. The authors limited inclusion criteria to prospective quantitative analyses of triggers for consultation for adult patients in the ED and inpatient settings since 2008. All retrospective studies, guidelines, reviews, case reports, editorials, and commentaries were excluded. Overall, threequarters of the studies were judged to be of moderate to high degree of bias, using the Newcastle-Ottawa scale for nonrandomized studies and the Cochrane risk of bias tool for randomized controlled trials. They concluded that nurses are frequently the primary operators of the trigger tools, and the primary categories focus on advanced illness states and the need for goals of care clarification. Future randomized controlled trials of high quality are still needed to define best practices and to distinguish primary from specialty PC needs. While the ED is included as a primary setting in this systematic review, only one study met their rigorous inclusion criteria. That study, published in 2008 by Mahony et al., demonstrated a low risk of bias [55].

In 2016, previous to the abovementioned systematic review, George et al. published a different systematic review of screening and assessment specific to the ED. Seven studies were included that met their criteria, which included the intervention of "a PC screening tool, assessment, or referral modality aimed at identifying patients appropriate for PC" [56].

The early studies utilize trained professionals within the ED to screen for PC patients with unmet needs. The earliest of these is from 2001. This study created a case finding strategy for at-risk patients >65 years old using a triage risk screening tool by the ED nurses, followed by a geriatric assessment tool performed by a geriatric clinical nurse specialist. While not specifically identifying a PC purpose, this study was successful in increasing referrals for supportive services to these elderly patients with unmet needs by sixfold during the 18-month study period and was able to demonstrate modest reductions (2-7%) in repeat ED use at three of four sites within 30 days [57]. Two later studies evaluated unmet PC needs for patients ≥ 65 years old with the goal of increasing early PC consultation. The first, by Mahony et al. and mentioned in both systematic reviews, used PC nurse practitioners. The second, by Glajchen et al., used a PC social worker. Both studies applied advanced illness criteria with needs assessments to determine which patients would likely benefit from early consultation. In the first study, of 291 patients who had needs assessments completed by the nurse practitioner, 90% were admitted to the hospital, most did not have advance directives, and a median of 68% died within 6 months. Of the deceased, those enrolled in hospice were more likely to die in their place of residence [55]. The second study reported on a group of patients identified by the social worker as having high unmet PC needs and therefore presumed high likelihood of benefit for early PC consultation. Among those recommended for PC consultation in the ED, only half formally received PC consultation, yet 80% died within the 8 months of study enrollment. The biggest reason for not receiving a PC consultation was reluctance on the part of the primary treating physician or the family [58]. Subsequent studies began to integrate the screening process into ED practice.

Speed

In 2011, the Screen for Palliative and End-of-life care needs in the Emergency Department (SPEED) tool was validated as a multidimensional system assessment tool to identify cancer patients on ED arrival who would benefit from PC consultation [59]. This tool was adapted and simplified into a five-question tool (5-SPEED) and integrated into an ED-based palliative intervention program as part of the initial bedside nurse assessment [60].

P-CaRES

In 2015, a broader, yet straightforward tool was developed to identify any patient who might benefit from PC consultation. First, content validity was established using a modified Delphi technique with experts from hospice and palliative medicine [12]. Next, using surveys and case vignettes, the tool demonstrated acceptability and reliability with emergency physicians in the Palliative Care and Rapid Emergency Screening (P-CaRES) project [61]. Finally, the tool was tested among emergency physicians in a convenience sample at a single institution over a 3-week period as an immediate look back on patients from their shift who were ≥ 65 years old. They found that 32% of patients screened positive for PC consultation, 70% of physicians found the screening tool acceptable to use, and the average time to completion was only 1.8 minutes per patient [62]. Most recently, as part of the Primary Palliative Care for Emergency Medicine (PRIM-ER) project, the P-CaRES tool was adapted into a clinical decision support tool to automate, within the electronic medical record system, the identification of and initiation of services for older adults with unmet PC needs [63].

The Surprise Question (SQ)

The question "would you be surprised if this patient died in the next 12 months?" has shown some utility as a simple tool to predict 1-year mortality. In the ED, the SQ has been used as a possible identifier of patients who might benefit from early PC consultation. In a 2018 study, ED physicians were asked the SQ in a convenience sample of 207 patients \geq 65 years old. The SQ alone had a sensitivity of 77% and specificity of 56% in predicting 1-year mortality [64]. Another study included all patients entering through the ED with advanced cancer over a 14-month period. Physicians were asked the SQ retrospectively, and if responding "would not be surprised," they also asked for an estimate of the Eastern Cooperative Oncological Group (ECOG) functional scale. Among 245 patients studied, the median survival was only 3 months. The SQ alone had a sensitivity of 89% in predicting 1-year mortality. The combination of the SQ and ECOG decreased sensitivity to 40% but increased specificity to 92% [65]. The SQ was also used to assess prognostic accuracy in heart failure patients in the

ED and exhibited a sensitivity of 79% and specificity of 57% [66]. Two recent studies adapted the SQ to "would you be surprised if the patient died in one month?" One assessed the prognostic value in all patients \geq 65 years old (sensitivity 20%, specificity 93%) [67], and the second assessed its prediction for in-hospital mortality (sensitivity 32%, specificity 85%) [68]. Overall the SQ shows promise as a simple and easily adaptable tool to help in identifying patients with unmet PC needs.

Other ED Screening Projects

Several other trigger-based efforts have been used to identify patients in the ED who would likely benefit from PC consultation. In one, the Functional Assessment Staging (FAST) criteria of >7 was used to identify patients \geq 70 years old with advanced dementia and found a disconnect between the predetermined criteria for PC consultation and the ED physician interpretation of appropriateness for consultation [69]. Another used the National Comprehensive Cancer Network (NCCN) guidelines to identify patients with advanced cancer, resulting in earlier PC consultation [70].

Overall, several screening tools appear to be helpful in identifying patients in the ED who may benefit from palliative approaches to care. Challenges remain for efficient integration of PC and in distinguishing primary from specialty-level needs.

Outcomes Research

For the purposes of this discussion, outcomes research refers to ED-initiated PC interventions and patient- and familyreported outcomes, healthcare utilization, and survival. Though studies exist dating from the 1990s, the majority were published in the last 5 years and include several published protocols for ongoing, large multicenter trials [71, 72]. The studies examine the impact of integrated PC models and differing strategies for clinical integration as described by the Improving Palliative Care in Emergency Medicine (IPAL-EM) project from the Center to Advance Palliative Care (CAPC). These include the following models: traditional consult, basic integration, advanced integration, and ED-focused advanced integration [7, 73].

There are two systematic reviews, the first published in 2016 by da Silva Soares et al., followed by Wilson et al. in 2020. The 2016 article considered only studies involving at least one member of a PC team and used hospital admission as their primary outcome. It found insufficient evidence to support the effect of ED-based PC interventions on the outcomes of interest, with the exception of one study suggesting a reduction in hospital length of stay. They urged efforts to

provide better evidence in the future through adequately powered and well-conducted randomized controlled trials [74]. The 2020 systematic review reported on the efficacy of ED-based palliative interventions on various outcomes. They noted that the existing evidence was conflicting with marked heterogeneity in design, population, and investigated outcomes, thus limiting a high-level analysis. Nonetheless, they concluded that compared with usual care, ED-initiated PC improves quality of life without decreasing survival time and likely decreases healthcare utilization metrics, such as length of stay [75].

Traditional Model of Consultation

The traditional model of consultation involves contacting a specialist to assist in patient care. Three particular studies show evidence of benefit from PC specialist consultation, focused primarily on the timing of consult. The first, from 2016, found that ED-initiated referral for advanced cancer patients led to earlier symptom control and decreased hospital length of stay compared to referrals initiated while inpatient [76]. A second study, from 2019, looked at adults presenting from long-term care facilities who were triaged as Emergency Severity Index Level 1 (most urgent). They found that formal PC involvement resulted in less aggressive care in 85% of cases; however most of these interventions occurred late in the hospital course with only 9.1% occurring while in the ED. They concluded that this was a missed opportunity to ensure goal-concordant care earlier in the hospital course [77]. A third study from the same year compared the in-hospital mortality of cancer patients directly admitted to the hospital versus those admitted from the ED. The ED patients had a higher overall in-hospital mortality, but PC consults within the first 3 days of admission were associated with decreased mortality when compared to later consultation. This supported a benefit to early PC consultation for ED patients with cancer and urgentlevel severity of illness [78].

Basic Integration

Basic integration describes ED and PC programs working collectively to meet objectives with at least some focused initiatives [7]. A randomized controlled trial from a single center in 2016 compared ED-initiated PC consultation to usual care for advanced cancer patients. Quality of life was improved without shortening survival, but with no clear impact on healthcare utilization metrics [79]. Another study from 2016 investigated the feasibility of an educational intervention to identify patients with less than 6 months prognosis. Of the patients identified, 91% of the patients met

hospice eligibility criteria, which succeeded in substantially increasing the number of days enrolled in hospice from baseline [80]. A second feasibility study from 2019 in Singapore tested a care model for metastatic cancer patients. It was designed as a tripartite ED-PC-hospice collaboration, which succeeded in earlier access to PC while, in the ED, including direct admission to a PC unit and to comfort care rooms [81].

Advanced Integration

Studies focused on advanced integration involve targeted screening criteria to identify at-risk patients and a higher level of ED engagement in designing protocols and providing care. The majority of these studies were of cohort design and contained educational interventions to improve general PC skills performed in conjunction with efforts to increase collaboration between emergency and PC providers. In Ireland, Tiernan et al. instituted a four-part intervention including a flagging system for patients known to the service, a screening checklist, education, as well as proactive PC engagement and found patients were 10.5 times more likely to receive a PC consult while in the ED after the intervention with subsequent decreased healthcare utilization [82]. The last study by Weng et al. from Taiwan describes a multi-stage intervention with strong educational components, including an inpatient hospice rotation for the residents, monthly ED interdisciplinary team meetings, and a dedicated cell phone app facilitating communication. This resulted in an increase in DNR orders and hospice consults from the ED [83].

ED-Focused Advanced Integration

Three studies describe ED-focused advanced integration efforts. An organizational level case report by Rosenberg et al. described the institutional development of an ED-based PC team which resulted in increased frequency of DNR orders, improved patient satisfaction, and decreased cost, without impacting the inpatient PC team's consultation volume [84]. A complex multi-step quality improvement initiative by Gruzden et al., the Geriatric Emergency Department Innovations in Care through Workforce, Informatics, and Structural Enhancements (GEDI WISE) model, implemented workforce enhancements, including role redefinition and education of existing providers, and was associated with expedited PC referrals and decreased geriatric ICU admissions from the ED as compared to baseline [85]. Finally, Liberman et al. successfully employed an advanced illness management program, which involved screening by an ED case manager followed by primarily ED-led goals of care conversations and PC available for support. They found

an increase in home hospice discharges compared to baseline (39.3% vs 0% respectively) [86].

The remaining outcomes studies focus on improving endof-life care and communication strategies.

End-of-Life Care

There is, in general, a lack of high-tier evidence regarding best practices for optimization of end-of-life care, specifically for ED patients. Many of the current practice recommendations for emergency care are based on evidence from other settings. A European observational study by Sedillot et al. suggested a five-step protocol for ventilator withdrawal, and two case reports examine end-of-life care transition [87]. A 2013 study by Shlamovitz et al. described improved symptom management after using ketamine for palliative sedation in an ED patient [88]. Wang suggested a systematic approach for comfort care transitions based on an extensive literature review [89]. A descriptive study by Economos et al. evaluated the appropriateness of end-of-life care provided to decedent ED patients based on published guidelines from the French Agency for accreditation and assessment, concluding that the majority of emergency providers recognize when a patient is actively dying and adjust treatments accordingly, with decreased provision of non-beneficial care [90]. Chor et al. described implementation of a protocolized care bundle aimed at reducing variability in the quality of end-of-life care received in a single institution in Singapore. They found that 74% of patients received opioids for pain and dyspnea and over 90% were allowed visitors; however the study lacked comparison data [91]. Finally, a 2020 randomized controlled crossover trial from Thailand compared high-flow nasal cannula to conventional oxygen therapy for relief of dyspnea and hypoxemia respiratory failure and showed significantly decreased dyspnea in the high-flow nasal cannula group compared to conventional therapy [92].

Communication

Best practices for ED-based goals of care conversations remain unclear, but a few studies have provided structure for future consideration. The first, a small Australian study evaluating the implementation of a form designed to guide and document conversations, found that while most physicians supported having the form, recommendations for its ideal format and content varied [93]. A subsequent systematic review found a similar lack of consensus with goals of care conversations in identifying patient candidates, as well as the content and documentation of these conversations [94]. Brief negotiated interview techniques have been successful for other patient care concerns in the ED, and a growing interest is developing in using these techniques as a structured approach to improve serious illness conversations in the ED [95, 96].

Resource Allocation

In this section, we examine studies describing the interface between emergency care and available internal and external support services. While many studies use decreased ED attendance as a surrogate metric for improved quality of care, we focus here primarily on studies describing collaborative efforts with the ED through improved care coordination or dedicated resources. Organizationally, these studies can be grouped by the timing and location of the resource in relation to the ED visit: (1) community-based care before or after ED attendance, including work done in the field of emergency medical services, or (2) hospital-based resources provided either during ED attendance or after admission. Quality improvement research predominates with most having a descriptive design.

Community-Based

Improved access to community-based PC services can lead to a significant reduction in ED use. In a scoping review of initiatives to reduce inappropriate or non-beneficial hospital admissions at end of life, Taylor et al. note that eight out of ten included studies observe a reduction in ED utilization for patients receiving community-based services [97]. This was also observed by Wiese et al. in a 2008 German crosssectional study examining emergency calls by cancer patients/families in the last 6 months of life [98]. Similar reductions in patterns of emergency service utilization were found in a 2016 US case control study on the impact of hospice services [99], a 2017 Canadian cohort study comparing palliative intent versus standard homecare nursing [100], and a 2018 Finnish study describing the impact of a growing outpatient PC clinic [101]. This last study also noted improved quality of documented advance care plans in palliative clinic patients.

The first interface for many palliative patients with emergency care is through emergency medical services. Several descriptive studies illustrate the potential benefits of close collaboration between EMS and palliative resources. A case report by Lamba et al. presents four case scenarios highlighting the need for improved collaboration and offers suggestions such as identifying PC champions and creating an action plan via an EMS dashboard [102]. A multicenter German cohort study by Wiese et al. found that emergency providers who were experienced in PC transported significantly fewer patients to the ED [103]. Montgomery et al.

describe a quality improvement initiative between EMS and community care providers in Edmonton allowing 61% of calls initiated for symptom crisis to be cared for at home when they would have traditionally required transfer [104]. Two qualitative studies by Kamphausen et al. and Patterson et al. note the impact of resource availability on avoiding unwanted transfer, particularly the importance of up-to-date documentation [105, 106]. There is no conclusive evidence on the best mechanism for communicating care preference across care settings. A small study by McQuown et al. on the transport of DNR orders with patients from extended care facilities found that while 68% of patients had DNR order, only 28% were transferred with the patient and not all of those were recognized by the receiving providers [107]. Lakin et al. found that while emergency physicians perceived advance care planning documents to be very useful, only 31% were very confident they could locate them in the electronic medical record [108]. The most studied method of communicating advance care plans is likely the physician order for life-sustaining treatment (POLST). However, a recent cohort study by Vranas et al. found that the majority of completed POLST forms had no impact on treatment intensity as they were often out of date, were not accessed, or did not indicate a desire for treatment limitations, thus limiting their utility [109].

Within the ED

In many healthcare systems, specialty PC services are a limited resource, and evidence on how to best expand care for emergency patients is also fairly limited. A cohort study by Purdy et al. demonstrated decreased ED attendance and inhospital deaths after implementing a care initiative including two hospital-based nurses whose role included identifying appropriate patients and facilitating discharge [110]. Lafond et al. described an institutional effort to improve care of terminal ED patients by allocating an ICU comfort care bed to facilitate rapid transfer. This resulted in a reduction in the number of patients dying on stretchers in the ED [111]. A cohort study by Liberman et al. demonstrated reduced 30-day re-visit and admission rates by embedding a specialist care coordinator in the ED who identified patients to be safely discharged with community resources [112].

Due to the events of the COVID-19 pandemic, there has been a surge of renewed interest in allocation of PC resources to include the ED. Several articles describing institutional responses appeared in the literature only months after the pandemic was declared. Chidiac et al. from the UK describe service capacity development by targeted palliative education of the generalist workforce, use of symptom management guidelines, and close partnership with community hospices [113]. The University of Washington's contingency and crisis plan for the ED included an embedded PC specialist for screening and assistance with ED goals of care discussions [114]. Colombia also dedicated a PC team within the ED for early goals of care conversations, noting a reduction in patients desiring full aggressive medical therapy from 82.7% to 18.2% following intervention [115].

After the ED

Gruzden et al. published a protocol for a multicenter randomized control trial comparing the effectiveness of specialty outpatient PC versus nurse-led telephonic palliative care, which will be the first of its kind [116]. Several descriptive studies of bereavement programs show potential benefit. One study from 1999 observed decreased grief intensity in high-risk participants [117]. A more recent study from 2019 noted the positive impact of a bereavement program also on clinicians [118].

Education for Clinicians

Death Disclosure

Early efforts to improve PC skills in the ED focused on the difficult and ubiquitous experience of performing a death disclosure. In 2002, Quest et al. used a structured curriculum for 16 evenly split first- and second-year EM residents to teach and practice death disclosure. The intervention was a combination of large group and small group didactics, followed by a standardized patient (SP) experience. While there was positive feedback and improved comfort with the intervention, the competency assessments of the residents were concordant between faculty and SP, but discordant with the resident self-assessments. Resident self-assessments of competency tended to be higher than faculty or SP assessment [119]. The same assessment tool, identified as the affective competency score (ACS), was later used to evaluate fourth year medical students rotating in the ED. It again demonstrated concordance with faculty and SP, but similar discordance with student self-assessment. Together, these findings support the use of the ACS as a valid and reliable tool for faculty and SPs to assess death disclosure [120].

A similar approach using the aptly named GRIEV_ING mnemonic was studied first with residents, then with paramedics. In the first study, 20 residents received a 2-hour educational session, including small group, role-play, and a didactic lecture. The residents were followed to assess retention after 3 months. Assessments of self-confidence and competency (by a trained SP and blinded audio rater, with high inter-rater reliability) improved from before and after the intervention with stability over time [121]. The GRIEV- ING mnemonic was later studied with 30 paramedics, limiting assessments to pre-/post-intervention, and found again an increase in self-confidence and competency (assessed by a trained SP) [122].

EPEC-EM

The Education in Palliative and End-of-life Care for Emergency Medicine (EPEC-EM) project is a comprehensive curriculum designed to facilitate teaching primary PC skills to EM clinicians. In 2010, Gisondi et al. published an adaptation of the curriculum for EM residents at a single academic medical center. The 14 curricular modules were divided into equal parts of synchronous (lectures) and asynchronous (electronic media) teaching strategies and demonstrated similar knowledge transfer, supporting selectively flexible approaches to delivering the curriculum [123]. The same year, Ponce et al. demonstrated improved comfort with prehospital personnel in a pilot study using the EPEC-EM modules as a training method for death disclosure and family-witnessed resuscitation [124]. A later study used selective topics from EPEC-EM to study the effect of EM resident behaviors and retention of information over a 6-month period after a brief educational intervention (4 hours of didactic learning in a single day). Based on comparison survey results, residents did not retain information on opioid conversions and dyspnea in the dying patient but retained knowledge of hospice qualifying diagnoses. This resulted in an observed 88% increase in direct ED to hospice referrals during the study period [125]. A more recent study used a simulation-based intervention focusing on rapid assessment of PC patients in the ED, derived from an EPEC-EM module. Residents demonstrated an increase in both the perceived importance of primary PC skills for EM physicians and self-reported confidence in using these skills [126]. In both of these last two studies, time limitations were identified as possible barriers to clinical implementation.

Training Efforts

Taking a broad view of how EM training programs in the United States incorporate PC principles, Kraus et al. surveyed the leadership from over 100 programs, representing 6 states where fully half of residency programs are located. They published in 2016 that a slight majority of EM programs taught PC skills. The following factors were associated with PC teaching: familiarity with EPEC-EM, HPM consult availability in the ED, and access to an HPM rotation. The topics of greatest importance were crucial conversations, pain management, and care of the imminently dying. The most effective teaching methods were

bedside teaching, HPM faculty mentorship, and case-based simulation. An interesting disconnect was noted between senior resident self-perception of competence and the following topics: the dying child, withdrawing and withholding life-sustaining treatments, and ethical and legal considerations of care. Faculty and resident barriers to implementation of the curriculum included a lack of subject expertise and a lack of interest [127]. By comparison, a Canadian study surveyed 26 programs (out of 36) and found that a minority of programs had any structured curriculum in PC. Of those, 100% were lecture or seminarbased approaches. Half had an elective rotation in HPM available, but only one had a mandatory rotation. Identified barriers to the incorporation of PC skills into the curriculum included lack of time and concerns of assuring that the content was relevant and current [128].

Two additional efforts deserve mention. The first used experts from HPM and EM to define content areas and competencies for primary PC skills and applied their recommendations to the EM Milestones for residency training [129]. The second provides the protocol of an expansive project currently underway looking at the impact of a comprehensive program of education and skills in primary PC for ED providers: the Primary Palliative Care for Emergency Medicine (PRIM-ER). This project will lead to a randomized controlled study of the effect of their interventions in 33 EDs across the country [71].

Summary

In this chapter, we have presented an overview of the rapidly growing research addressing the interface of PC within the ED. For structure, we relied on guidance from a 2011 workgroup of experts from both fields who created six priorities for research efforts: descriptive, attitudinal, screening, outcomes, resource allocation, and education for clinicians. Further categorization used study descriptions from the methodology sections for each referenced article (see Table 63.1). While a true analysis of the hierarchy of evidence was beyond the scope of this chapter, recent robust and well-powered study designs show promise in the emerging literature to guide future best practices in the ED care of patients with serious illness.

References

- Chan GK. End-of-life and palliative care in the emergency department: a call for research, education, policy and improved practice in this frontier area. J Emerg Nurs. 2006;32(1):101–3.
- Quest TE, Asplin BR, Cairns CB, Hwang U, Pines JM. Research priorities for palliative and end-of-life care in the emergency setting. Acad Emerg Med. 2011;18(6):e70–6.

- 3. Brown J, Grudzen C, Kyriacou DN, Obermeyer Z, Quest T, Rivera D. The emergency care of patients with cancer: setting the research agenda. Ann Emerg Med. 2016;68(6):706–11.
- 4. Greenhalgh T. How to read a paper: the basics of evidence-based medicine and healthcare. Hoboken, NJ: Wiley Blackwell; 2019.
- Oulton J, Rhodes SM, Howe C, Fain MJ, Mohler MJ. Advance directives for older adults in the emergency department: a systematic review. J Palliat Med. 2015;18(6):500–5.
- Bone AE, Evans CJ, Etkind SN, Sleeman KE, Gomes B, Aldridge M, et al. Factors associated with older people's emergency department attendance towards the end of life: a systematic review. Eur J Pub Health. 2019;29(1):67–74.
- 7. Quest T, Herr S, Lamba S, Weissman D, Board I-EA. Demonstrations of clinical initiatives to improve palliative care in the emergency department: a report from the IPAL-EM Initiative. Ann Emerg Med. 2013;61(6):661–7.
- Grudzen CR, Richardson LD, Morrison M, Cho E, Morrison RS. Palliative care needs of seriously ill, older adults presenting to the emergency department. Acad Emerg Med. 2010;17(11):1253–7.
- Van Tricht M, Riochet D, Batard E, Martinage A, Montassier E, Potel G, et al. Palliative care for patients who died in emergency departments: analysis of a multicentre cross-sectional survey. Emerg Med J. 2012;29(10):795–7.
- Lamba S, Nagurka R, Walther S, Murphy P. Emergencydepartment-initiated palliative care consults: a descriptive analysis. J Palliat Med. 2012;15(6):633–6.
- 11. Blackwell RW, Lowton K, Robert G, Grudzen C, Grocott P. Using Experience-based Co-design with older patients, their families and staff to improve palliative care experiences in the emergency department: a reflective critique on the process and outcomes. Int J Nurs Stud. 2017;68:83–94.
- George N, Barrett N, McPeake L, Goett R, Anderson K, Baird J. Content Validation of a Novel screening tool to identify emergency department patients with significant palliative care needs. Acad Emerg Med. 2015;22(7):823–37.
- Köstenberger M, Neuwersch S, Weixler D, Pipam W, Zink M, Likar R. Prevalence of palliative care patients in emergency departments. Wien Klin Wochenschr. 2019;131(17–18):404–9.
- Chen B, Kanaan C, Jaiyesimi I, Ezekwudo D, Swor R. Clinical characteristics of patients with cancer presenting to the emergency department and their use of emergency medical service transport. Prehosp Emerg Care. 2020:1–9.
- Barbera L, Taylor C, Dudgeon D. Why do patients with cancer visit the emergency department near the end of life? CMAJ. 2010;182(6):563–8.
- Mayer DK, Travers D, Wyss A, Leak A, Waller A. Why do patients with cancer visit emergency departments? Results of a 2008 population study in North Carolina. J Clin Oncol. 2011;29(19):2683–8.
- Henson L, Gao W, Higginson I, Smith M, Davies J, Ellis-Smith C, et al. Emergency department attendance by patients with cancer in the last month of life: a systematic review and meta-analysis. Lancet. 2015;385(Suppl 1):S41.
- Elsayem AF, Merriman KW, Gonzalez CE, Yeung SC, Chaftari PS, Reyes-Gibby C, et al. Presenting symptoms in the emergency department as predictors of intensive care unit admissions and hospital mortality in a comprehensive cancer center. J Oncol Pract. 2016;12(5):e554–63.
- Rivera DR, Gallicchio L, Brown J, Liu B, Kyriacou DN, Shelburne N. Trends in adult cancer-related emergency department utilization: an analysis of data from the nationwide emergency department sample. JAMA Oncol. 2017;3(10):e172450.
- Henson LA, Higginson IJ, Gao W, BuildCARE. What factors influence emergency department visits by patients with cancer at the end of life? Analysis of a 124,030 patient cohort. Palliat Med. 2018;32(2):426–38.

- Caterino JM, Adler D, Durham DD, Yeung SJ, Hudson MF, Bastani A, et al. Analysis of diagnoses, symptoms, medications, and admissions among patients with cancer presenting to emergency departments. JAMA Netw Open. 2019;2(3):e190979.
- 22. Verhoef MJ, de Nijs EJM, Ootjers CS, Fiocco M, Fogteloo AJ, Heringhaus C, et al. End-of-life trajectories of patients with hematological malignancies and patients with advanced solid tumors visiting the emergency department: the need for a proactive integrated care approach. Am J Hosp Palliat Care. 2020;37(9):692–700.
- 23. Gaucher N, Humbert N, Gauvin F. What do we know about pediatric palliative care patients consulting to the pediatric emergency department? Pediatr Emerg Care. 2018; https://doi.org/10.1097/ PEC.000000000001620. Epub ahead of print.
- 24. Rocha AMO, de Mello MJG, Torres JRD, Valença NO, Maia ACA, Cavalcanti NV. Palliative care in congenital syndrome of the Zika virus associated with hospitalization and emergency consultation: palliative care and congenital syndrome of Zika. J Trop Med. 2018;2018:1025193.
- Ausband SC, March JA, Brown LH. National prevalence of palliative care protocols in emergency medical services. Prehosp Emerg Care. 2002;6(1):36–41.
- Carron PN, Dami F, Diawara F, Hurst S, Hugli O. Palliative care and prehospital emergency medicine: analysis of a case series. Medicine (Baltimore). 2014;93(25):e128.
- Wallace EM, Cooney MC, Walsh J, Conroy M, Twomey F. Why do palliative care patients present to the emergency department? Avoidable or unavoidable? Am J Hosp Palliat Care. 2013;30(3):253–6.
- Hjermstad MJ, Kolflaath J, Løkken AO, Hanssen SB, Normann AP, Aass N. Are emergency admissions in palliative cancer care always necessary? Results from a descriptive study. BMJ Open. 2013;3(5)
- 29. Kao YH, Liu YT, Koo M, Chiang JK. Factors associated with emergency services use in Taiwanese advanced cancer patients receiving palliative home care services during out-of-hours periods: a retrospective medical record study. BMC Palliat Care. 2018;17(1):46.
- Lipinski M, Eagles D, Fischer LM, Mielniczuk L, Stiell IG. Heart failure and palliative care in the emergency department. Emerg Med J. 2018;35(12):726–9.
- 31. Green E, Shaw SE, Harris T. 'They shouldn't be coming to the ED, should they?' A qualitative study of why patients with palliative care needs present to the emergency department. BMJ Support Palliat Care. 2019;9(4):e29.
- 32. Shin SH, Hui D, Chisholm GB, Kwon JH, San-Miguel MT, Allo JA, et al. Characteristics and outcomes of patients admitted to the acute palliative care unit from the emergency center. J Pain Symptom Manag. 2014;47(6):1028–34.
- 33. Spilsbury K, Rosenwax L, Arendts G, Semmens JB. The association of community-based palliative care with reduced emergency department visits in the last year of life varies by patient factors. Ann Emerg Med. 2017;69(4):416–25.
- 34. Wiese CH, Bartels UE, Ruppert D, Marung H, Luiz T, Graf BM, et al. Treatment of palliative care emergencies by prehospital emergency physicians in Germany: an interview based investigation. Palliat Med. 2009;23(4):369–73.
- 35. Grudzen CR, Richardson LD, Hopper SS, Ortiz JM, Whang C, Morrison RS. Does palliative care have a future in the emergency department? Discussions with attending emergency physicians. J Pain Symptom Manag. 2012;43(1):1–9.
- Lamba S, Nagurka R, Zielinski A, Scott SR. Palliative care provision in the emergency department: barriers reported by emergency physicians. J Palliat Med. 2013;16(2):143–7.
- 37. Weil J, Weiland TJ, Lane H, Jelinek GA, Boughey M, Marck CH, et al. What's in a name? A qualitative exploration of what is under-

stood by "palliative care" in the emergency department. Palliat Med. 2015;29(4):293–301.

- Côté AJ, Payot A, Gaucher N. Palliative care in the pediatric emergency department: findings from a qualitative study. Ann Emerg Med. 2019;74(4):481–90.
- 39. Smith AK, Fisher J, Schonberg MA, Pallin DJ, Block SD, Forrow L, et al. Am I doing the right thing? Provider perspectives on improving palliative care in the emergency department. Ann Emerg Med. 2009;54(1):86–93. e1
- Russ A, Mountain D, Rogers IR, Shearer F, Monterosso L, Ross-Adjie G, et al. Staff perceptions of palliative care in a public Australian, metropolitan emergency department. Emerg Med Australas. 2015;27(4):287–94.
- Wright RJ, Lowton K, Robert G, Grudzen CR, Grocott P. Emergency department staff priorities for improving palliative care provision for older people: a qualitative study. Palliat Med. 2018;32(2):417–25.
- 42. Grudzen CR, Richardson LD, Major-Monfried H, Kandarian B, Ortiz JM, Morrison RS. Hospital administrators' views on barriers and opportunities to delivering palliative care in the emergency department. Ann Emerg Med. 2013;61(6):654–60.
- Rivera MR, Torres FS. Lack of training and comfort level with provision of palliative care in Puerto Rican emergency departments. Bol Asoc Med P R. 2015;107(2):92–6.
- 44. Argintaru N, Quinn KL, Chartier LB, Lee J, Hannam P, O'Connor E, et al. Perceived barriers and facilitators to goals of care discussions in the emergency department: a descriptive analysis of the views of emergency medicine physicians and residents. CJEM. 2019;21(2):211–8.
- 45. Smith AK, Schonberg MA, Fisher J, Pallin DJ, Block SD, Forrow L, et al. Emergency department experiences of acutely symptomatic patients with terminal illness and their family caregivers. J Pain Symp Manag. 2010;39(6):972–81.
- 46. Cooper E, Hutchinson A, Sheikh Z, Taylor P, Townend W, Johnson MJ. Palliative care in the emergency department: a systematic literature qualitative review and thematic synthesis. Palliat Med. 2018;32(9):1443–54.
- 47. Di Leo S, Alquati S, Autelitano C, Costantini M, Martucci G, De Vincenzo F, et al. Palliative care in the emergency department as seen by providers and users: a qualitative study. Scand J Trauma Resusc Emerg Med. 2019;27(1):88.
- Fernández-Sola C, Cortés MM, Hernández-Padilla JM, Torres CJ, Terrón JM, Granero-Molina J. Defining dignity in end-of-life care in the emergency department. Nurs Ethics. 2017;24(1):20–32.
- 49. Giles TM, Hammad K, Breaden K, Drummond C, Bradley SL, Gerace A, et al. Nurses' perceptions and experiences of caring for patients who die in the emergency department setting. Int Emerg Nurs. 2019;47:100789.
- Alqahtani AJ, Mitchell G. End-of-life care challenges from staff viewpoints in emergency departments: systematic review. Healthcare (Basel). 2019;7(3):83.
- Mughal A, Evans C. Views and experiences of nurses in providing end-of-life care to patients in an ED context: a qualitative systematic review. Emerg Med J. 2020;37(5):265–72.
- Meo N, Hwang U, Morrison RS. Resident perceptions of palliative care training in the emergency department. J Palliat Med. 2011;14(5):548–55.
- Lamba S, Pound A, Rella JG, Compton S. Emergency medicine resident education in palliative care: a needs assessment. J Palliat Med. 2012;15(5):516–20.
- Kistler EA, Stevens E, Scott E, Philpotts LL, Greer JA, Greenwald JL. Triggered palliative care consults: a systematic review of interventions for hospitalized and emergency department patients. J Pain Symptom Manag. 2020;60(2):460–75.
- 55. Mahony SO, Blank A, Simpson J, Persaud J, Huvane B, McAllen S, et al. Preliminary report of a palliative care and case man-

agement project in an emergency department for chronically ill elderly patients. J Urban Health. 2008;85(3):443–51.

- 56. George N, Phillips E, Zaurova M, Song C, Lamba S, Grudzen C. Palliative care screening and assessment in the emergency department: a systematic review. J Pain Symptom Manag. 2016;51(1):108–19.e2.
- 57. Mion LC, Palmer RM, Anetzberger GJ, Meldon SW. Establishing a case-finding and referral system for at-risk older individuals in the emergency department setting: the SIGNET model. J Am Geriatr Soc. 2001;49(10):1379–86.
- Glajchen M, Lawson R, Homel P, Desandre P, Todd KH. A rapid two-stage screening protocol for palliative care in the emergency department: a quality improvement initiative. J Pain Symptom Manag. 2011;42(5):657–62.
- 59. Richards CT, Gisondi MA, Chang CH, Courtney DM, Engel KG, Emanuel L, et al. Palliative care symptom assessment for patients with cancer in the emergency department: validation of the screen for palliative and end-of-life care needs in the emergency department instrument. J Palliat Med. 2011;14(6):757–64.
- Reuter Q, Marshall A, Zaidi H, Sista P, Powell ES, McCarthy DM, et al. Emergency department-based palliative interventions: a novel approach to palliative care in the emergency department. J Palliat Med. 2019;22(6):649–55.
- Bowman J, George N, Barrett N, Anderson K, Dove-Maguire K, Baird J. Acceptability and reliability of a novel palliative care screening tool among emergency department providers. Acad Emerg Med. 2016;23(6):694–702.
- 62. Ouchi K, Block SD, Schonberg MA, Jamieson ES, Aaronson EL, Pallin DJ, et al. Feasibility testing of an emergency department screening tool to identify older adults appropriate for palliative care consultation. J Palliat Med. 2017;20(1):69–73.
- 63. Tan A, Durbin M, Chung FR, Rubin AL, Cuthel AM, McQuilkin JA, et al. Design and implementation of a clinical decision support tool for Primary Palliative Care for Emergency Medicine (PRIM-ER). BMC Med Inform Decis Mak. 2020;20(1):13.
- 64. Ouchi K, Jambaulikar G, George NR, Xu W, Obermeyer Z, Aaronson EL, et al. The "surprise question" asked of emergency physicians may predict 12-month mortality among older emergency department patients. J Palliat Med. 2018;21(2):236–40.
- 65. Verhoef MJ, de Nijs EJM, Fiocco M, Heringhaus C, Horeweg N, van der Linden YM. Surprise question and performance status indicate urgency of palliative care needs in patients with advanced cancer at the emergency department: an observational cohort study. J Palliat Med. 2019;
- 66. Aaronson EL, George N, Ouchi K, Zheng H, Bowman J, Monette D, et al. The surprise question can be used to identify heart failure patients in the emergency department who would benefit from palliative care. J Pain Symptom Manag. 2019;57(5):944–51.
- 67. Ouchi K, Strout T, Haydar S, Baker O, Wang W, Bernacki R, et al. Association of Emergency clinicians' assessment of mortality risk with actual 1-month mortality among older adults admitted to the hospital. JAMA Netw Open. 2019;2(9):e1911139.
- Haydar SA, Strout TD, Bond AG, Han PK. Prognostic value of a modified surprise question designed for use in the emergency department setting. Clin Exp Emerg Med. 2019;6(1):70–6.
- Ouchi K, Wu M, Medairos R, Grudzen CR, Balsells H, Marcus D, et al. Initiating palliative care consults for advanced dementia patients in the emergency department. J Palliat Med. 2014;17(3):346–50.
- Kistler EA, Sean Morrison R, Richardson LD, Ortiz JM, Grudzen CR. Emergency department-triggered palliative care in advanced cancer: proof of concept. Acad Emerg Med. 2015;22(2):237–9.
- 71. Grudzen CR, Brody AA, Chung FR, Cuthel AM, Mann D, McQuilkin JA, et al. Primary Palliative Care for Emergency Medicine (PRIM-ER): protocol for a pragmatic, clusterrandomised, stepped wedge design to test the effectiveness of pri-

mary palliative care education, training and technical support for emergency medicine. BMJ Open. 2019;9(7):e030099.

- 72. Yash Pal R, Kuan WS, Tiah L, Kumar R, Wong YKY, Shi L, et al. End-of-life management protocol offered within emergency room (EMPOWER): study protocol for a multicentre study. BMJ Open. 2020;10(4):e036598.
- 73. Lamba S, DeSandre PL, Todd KH, Bryant EN, Chan GK, Grudzen CR, et al. Integration of palliative care into emergency medicine: the Improving Palliative Care in Emergency Medicine (IPAL-EM) collaboration. J Emerg Med. 2014;46(2):264–70.
- 74. da Silva SD, Nunes CM, Gomes B. Effectiveness of emergency department based palliative care for adults with advanced disease: a systematic review. J Palliat Med. 2016;19(6):601–9.
- 75. Wilson JG, English DP, Owyang CG, Chimelski EA, Grudzen CR, Wong HN, et al. End-of-life care, palliative care consultation, and palliative care referral in the emergency department: a systematic review. J Pain Symptom Manage. 2020;59(2):372–83.e1.
- 76. Delgado-Guay MO, Rodriguez-Nunez A, Shin SH, Chisholm G, Williams J, Frisbee-Hume S, et al. Characteristics and outcomes of patients with advanced cancer evaluated by a palliative care team at an emergency center. A retrospective study. Support Care Cancer. 2016;24(5):2287–95.
- Mogul AS, Cline DM, Gabbard J, Bryant C. Missed opportunities: integrating palliative care into the emergency department for older adults presenting as level i triage priority from long-term care facilities. J Emerg Med. 2019;56(2):145–52.
- El Majzoub I, Qdaisat A, Chaftari PS, Yeung SJ, Sawaya RD, Jizzini M, et al. Association of emergency department admission and early inpatient palliative care consultation with hospital mortality in a comprehensive cancer center. Support Care Cancer. 2019;27(7):2649–55.
- Grudzen CR, Richardson LD, Johnson PN, Hu M, Wang B, Ortiz JM, et al. Emergency department-initiated palliative care in advanced cancer: a randomized clinical trial. JAMA Oncol. 2016;2(5):591–8.
- Highet BH, Hsieh YH, Smith TJ. A pilot trial to increase hospice enrollment in an inner city, academic emergency department. J Emerg Med. 2016;51(2):106–13.
- Koh MYH, Lee JF, Montalban S, Foo CL, Hum AYM. ED-PALS: a comprehensive palliative care service for oncology patients in the emergency department. Am J Hosp Palliat Care. 2019;36(7):571–6.
- 82. Tiernan E, Ryan J, Casey M, Hale A, O'Reilly V, Devenish M, et al. A quasi-experimental evaluation of an intervention to increase palliative medicine referral in the emergency department. J Health Serv Res Policy. 2019;24(3):155–63.
- 83. Weng TC, Yang YC, Chen PJ, Kuo WF, Wang WL, Ke YT, et al. Implementing a novel model for hospice and palliative care in the emergency department: an experience from a tertiary medical center in Taiwan. Medicine (Baltimore). 2017;96(19):e6943.
- Rosenberg M, Rosenberg L. Integrated model of palliative care in the emergency department. West J Emerg Med. 2013;14(6):633–6.
- 85. Grudzen C, Richardson LD, Baumlin KM, Winkel G, Davila C, Ng K, et al. Redesigned geriatric emergency care may have helped reduce admissions of older adults to intensive care units. Health Aff (Millwood). 2015;34(5):788–95.
- 86. Liberman T, Kozikowski A, Kwon N, Emmert B, Akerman M, Pekmezaris R. Identifying advanced illness patients in the emergency department and having goals-of-care discussions to assist with early hospice referral. J Emerg Med. 2018;54(2):191–7.
- 87. Sedillot N, Holzapfel L, Jacquet-Francillon T, Tafaro N, Eskandanian A, Eyraud S, et al. A five-step protocol for withholding and withdrawing of life support in an emergency department: an observational study. Eur J Emerg Med. 2008;15(3):145–9.

- Shlamovitz GZ, Elsayem A, Todd KH. Ketamine for palliative sedation in the emergency department. J Emerg Med. 2013;44(2):355–7.
- Wang D, Creel-Bulos C. A systematic approach to comfort care transitions in the emergency department. J Emerg Med. 2019;56(3):267–74.
- Economos G, Cavalli P, Guérin T, Filbet M, Perceau-Chambard E. Quality of end-of-life care in the emergency department. Turk J Emerg Med. 2019;19(4):141–5.
- Chor WPD, Wong SYP, Ikbal MFBM, Kuan WS, Chua MT, Pal RY. Initiating end-of-life care at the emergency department: an observational study. Am J Hosp Palliat Care. 2019;36(11):941–6.
- 92. Ruangsomboon O, Dorongthom T, Chakorn T, Monsomboon A, Praphruetkit N, Limsuwat C, et al. High-flow nasal cannula versus conventional oxygen therapy in relieving dyspnea in emergency palliative patients with do-not-intubate status: a randomized crossover study. Ann Emerg Med. 2020;75(5):615–26.
- 93. Mills AC, Levinson M, Dunlop WA, Cheong E, Cowan T, Hanning J, et al. Testing a new form to document 'goals-of-care' discussions regarding plans for end-of-life care for patients in an Australian emergency department. Emerg Med Australas. 2018;30(6):777–84.
- Hanning J, Walker KJ, Horrigan D, Levinson M, Mills A. Review article: goals-of-care discussions for adult patients nearing end of life in emergency departments: a systematic review. Emerg Med Australas. 2019;31(4):525–32.
- Ouchi K, George N, Revette AC, Hasdianda MA, Fellion L, Reust A, et al. Empower seriously ill older adults to formulate their goals for medical care in the emergency department. J Palliat Med. 2019;22(3):267–73.
- 96. Leiter RE, Yusufov M, Hasdianda MA, Fellion LA, Reust AC, Block SD, et al. Fidelity and feasibility of a brief emergency department intervention to empower adults with serious illness to initiate advance care planning conversations. J Pain Symptom Manag. 2018;56(6):878–85.
- 97. Taylor R, Ellis J, Gao W, Searle L, Heaps K, Davies R, et al. A scoping review of initiatives to reduce inappropriate or nonbeneficial hospital admissions and bed days in people nearing the end of their life: much innovation, but limited supporting evidence. BMC Palliat Care. 2020;19(1):24.
- Wiese CH, Bartels U, Ruppert D, Marung H, Graf BM, Hanekop GG. [Emergency outpatient palliative care in acute situations by paramedics]. Wien Klin Wochenschr. 2008;120(17–18):539–46. (*Article in German*).
- 99. Obermeyer Z, Clarke AC, Makar M, Schuur JD, Cutler DM. Emergency care use and the medicare hospice benefit for individuals with cancer with a poor prognosis. J Am Geriatr Soc. 2016;64(2):323–9.
- 100. Sutradhar R, Barbera L, Seow HY. Palliative homecare is associated with reduced high- and low-acuity emergency department visits at the end of life: a population-based cohort study of cancer decedents. Palliat Med. 2017;31(5):448–55.
- 101. Hirvonen OM, Alalahti JE, Syrjänen KJ, Jyrkkiö SM. End-of-life decisions guiding the palliative care of cancer patients visiting emergency department in South Western Finland: a retrospective cohort study. BMC Palliat Care. 2018;17(1):128.
- 102. Lamba S, Schmidt TA, Chan GK, Todd KH, Grudzen CR, Weissman DE, et al. Integrating palliative care in the out-ofhospital setting: four things to jump-start an EMS-palliative care initiative. Prehosp Emerg Care. 2013;17(4):511–20.
- 103. Wiese CH, Bartels UE, Marczynska K, Ruppert D, Graf BM, Hanekop GG. Quality of out-of-hospital palliative emergency care depends on the expertise of the emergency medical team– a prospective multi-centre analysis. Support Care Cancer. 2009;17(12):1499–506.

- 104. Montgomery CL, Pooler C, Arsenault JE, Berean C, Sharman R, Cameron CL, et al. Innovative urgent care for the palliative patient at home. Home Healthc Now. 2017;35(4):196–201.
- 105. Kamphausen A, Roese H, Oechsle K, Issleib M, Zöllner C, Bokemeyer C, et al. Challenges faced by prehospital emergency physicians providing emergency care to patients with advanced incurable diseases. Emerg Med Int. 2019;2019:3456471.
- 106. Patterson R, Standing H, Lee M, Dalkin S, Lhussier M, Exley C, et al. Paramedic information needs in end-of-life care: a qualitative interview study exploring access to a shared electronic record as a potential solution. BMC Palliat Care. 2019;18(1):108.
- 107. McQuown CM, Frey JA, Amireh A, Chaudhary A. Transfer of DNR orders to the ED from extended care facilities. Am J Emerg Med. 2017;35(7):983–5.
- 108. Lakin JR, Isaacs E, Sullivan E, Harris HA, McMahan RD, Sudore RL. Emergency physicians' experience with advance care planning documentation in the electronic medical record: useful, needed, and elusive. J Palliat Med. 2016;19(6):632–8.
- 109. Vranas KC, Lin AL, Zive D, Tolle SW, Halpern SD, Slatore CG, et al. The association of physician orders for life-sustaining treatment with intensity of treatment among patients presenting to the emergency department. Ann Emerg Med. 2020;75(2):171–80.
- 110. Purdy S, Lasseter G, Griffin T, Wye L. Impact of the Marie Curie Cancer Care Delivering Choice Programme in Somerset and North Somerset on place of death and hospital usage: a retrospective cohort study. BMJ Support Palliat Care. 2015;5(1):34–9.
- 111. Lafond P, Chalayer E, Roussier M, Weber E, Lacoin-Reynaud Q, Tardy B. A hospice and palliative care bed dedicated to patients admitted to the emergency department for end-of-life care. Am J Hosp Palliat Care. 2016;33(4):403–6.
- 112. Liberman T, Roofeh R, Sohn N, Brave M, Smith A, Willis H, et al. The GAP-ED Project: improving care for elderly patients presenting to the emergency department. J Emerg Med. 2020;58(2):191–7.
- Chidiac C, Feuer D, Naismith J, Flatley M, Preston N. Emergency palliative care planning and support in a covid-19 pandemic. J Palliat Med. 2020;23(6):752–3.
- 114. Fausto J, Hirano L, Lam D, Mehta A, Mills B, Owens D, et al. Creating a palliative care inpatient response plan for COVID-19-The UW medicine experience. J Pain Symptom Manag. 2020;60(1):e21–e6.
- 115. Lee J, Abrukin L, Flores S, Gavin N, Romney ML, Blinderman CD, Nakagawa S. Early intervention of palliative care in the emergency department during the COVID-19 pandemic. JAMA Intern Med. 2020;e202713.
- 116. Grudzen CR, Shim DJ, Schmucker AM, Cho J, Goldfeld KS, Investigators E. Emergency Medicine Palliative Care Access (EMPallA): protocol for a multicentre randomised controlled trial comparing the effectiveness of specialty outpatient versus nurseled telephonic palliative care of older adults with advanced illness. BMJ Open. 2019;9(1):e025692.
- 117. Ting SM, Li P, Lau FL, Chan YC, Chan YK, Lai PW, et al. Acute bereavement care in the emergency department: does the professional-supported volunteers model work? Eur J Emerg Med. 1999;6(3):237–43.
- 118. Cooper JJ, Stock RC, Wilson SJ. Emergency department grief support: a multidisciplinary intervention to provide bereavement support after death in the emergency department. J Emerg Med. 2020;58(1):141–7.
- 119. Quest TE, Otsuki JA, Banja J, Ratcliff JJ, Heron SL, Kaslow NJ. The use of standardized patients within a procedural competency model to teach death disclosure. Acad Emerg Med. 2002;9(11):1326–33.
- Quest TE, Ander DS, Ratcliff JJ. The validity and reliability of the affective competency score to evaluate death disclosure using standardized patients. J Palliat Med. 2006;9(2):361–70.

- 121. Hobgood C, Harward D, Newton K, Davis W. The educational intervention "GRIEV_ING" improves the death notification skills of residents. Acad Emerg Med. 2005;12(4):296–301.
- 122. Hobgood C, Mathew D, Woodyard DJ, Shofer FS, Brice JH. Death in the field: teaching paramedics to deliver effective death notifications using the educational intervention "GRIEV_ING". Prehosp Emerg Care. 2013;17(4):501–10.
- 123. Gisondi MA, Lu DW, Yen M, Norris R, Courtney DM, Tanabe P, et al. Adaptation of EPEC-EM curriculum in a residency with asynchronous learning. West J Emerg Med. 2010;11(5):491–9.
- 124. Ponce A, Swor R, Quest TE, Macy M, Meurer W, Sasson C. Death notification training for prehospital providers: a pilot study. Prehosp Emerg Care. 2010;14(4):537–42.
- 125. DeVader TE, Jeanmonod R. The effect of education in hospice and palliative care on emergency medicine residents' knowledge and referral patterns. J Palliat Med. 2012;15(5):510–5.

- Goldonowicz JM, Runyon MS, Bullard MJ. Palliative care in the emergency department: an educational investigation and intervention. BMC Palliat Care. 2018;17(1):43.
- 127. Kraus CK, Greenberg MR, Ray DE, Dy SM. Palliative Care education in emergency medicine residency training: a survey of program directors, associate program directors, and assistant program directors. J Pain Symptom Manag. 2016;51(5):898–906.
- 128. Baylis J, Harris DR, Chen C, Ting DK, Clark K, Kwan A, et al. Palliative and end-of-life care education in Canadian emergency medicine residency programs: a national cross-sectional survey. CJEM. 2019;21(2):219–25.
- 129. Shoenberger J, Lamba S, Goett R, DeSandre P, Aberger K, Bigelow S, et al. Development of hospice and palliative medicine knowledge and skills for emergency medicine residents: using the Accreditation Council for Graduate Medical Education Milestone Framework. AEM Educ Train. 2018;2(2):130–45.

Part VI

Contextual



Kenneth V. Iserson



64

Introduction

Bioethical issues often arise when treating emergency department (ED) and prehospital care patients. Actual or anticipated bioethical dilemmas commonly occur among patients with hematologic and oncologic diseases, and these dilemmas may require slightly different approaches than in other ED patients due to attitudes toward and the nature of the disease processes. Bioethical dilemmas raised by emergency hematologic-oncologic patients fall into four categories (Table 64.1): decision-making, treatment demands and refusals, system problems, and notifications.

Bioethics can be a nebulous concept, so the first order of business will be to lay the groundwork by describing bioethics and discussing how it fits into our societal and professional value systems. Then, I will briefly review basic ethical (foundational) theories and the methods used to think through ethical dilemmas, followed by a discussion of the mid-level ethical principles with which clinicians may be more familiar. While they may appear superficial or oversimplified, these mid-level principles provide an easy way to think about the issues posed in bioethical dilemmas and policy development. Therefore, when treating emergency patients with hematologic and oncologic illnesses, we use them to convey common moral themes, such as decisionmaking, demands for and refusals of treatment, and system constraints. Finally, I will move into the area of virtues to discuss notifications to patients and survivors.

How Bioethics Fits into Our Societal and Professional Value Systems

Bioethics, or clinical ethics, describes how we apply professional and societal values in an organized way to find rea
 Table 64.1
 Categories of bioethical issues encountered when working with patients with hematologic-oncologic diseases and their families

Decision-making (autonomy)

- 1. Dying. Surrogates and advance directives (living will, medical power of attorney, prehospital advance directives)
- 2. Decision-making capacity
- Treatment demands/refusals (*beneficence*, *nonmaleficence*)
- 1. Demands to "do everything"
- 2. Palliative care decisions (demand to do "nothing"—Comfort care only)
- 3. Refusal of analgesics
- 4. Refusal of possibly beneficial treatment (including decisions based on religious beliefs)

System dilemmas (distributive justice, confidentiality)

- System problems (inability to pay, intentional/unintentional release of patient information, undocumented alien, "wrong" insurance or medical system/group)
- 2. Collegial problems (refusal to see patient, abandonment, etc.)
- 3. Research protocols
- Notification (honesty with sensitivity)
 - 1. Notifying patient/family of diagnosis
- 2. Notifying survivors of death

soned and defensible solutions for moral dilemmas. Moral dilemmas are those situations in which an individual must make a decision between conflicting or competing values. The resolutions to such dilemmas, however, do not always hinge on determining right versus wrong or good versus evil. Rather, moral dilemmas more often deal with "gray areas," where the situations or resolutions initially seem to be equivalent, i.e., situations with seemingly equal merit or apparently equal injury. In these more ambiguous situations, we use ethical values to help determine a morally acceptable course of action.

In a pluralistic society, we derive these values from a variety of sources, including the general cultural, philosophical, and religious moral traditions, the social norms embodied in law, and our professional oaths and ethical codes. Each of these sources claims moral superiority. The goal of bioethics is to help us understand, interpret, and weigh these competing moral values [1].

K. V. Iserson (🖂)

Department of Emergency Medicine, University of Arizona, Tucson, AZ, USA e-mail: kvi@u.arizona.edu

K. H. Todd et al. (eds.), Oncologic Emergency Medicine, https://doi.org/10.1007/978-3-030-67123-5_64

Case Study 1

The following case represents an example of how bioethics can be applied when a cancer patient presents to the ED.

An exsanguinating adult leukemic patient, awake and still with medical decision-making capacity, arrives in the ED and explicitly states that, owing to long-standing religious beliefs, she wants no blood or blood products. The physician, with a professional duty and moral commitment to preserve life, does not personally agree with the patient's decision. Yet, society (through the benchmark of court decisions) has repeatedly sided with the patient's right to refuse such treatment.

In this case, the patient's autonomy and right to practice her religion are recognized as the overriding values. The case becomes somewhat less clear when the patient lacks decision-making capacity, is a minor, or appears to be under external pressures (such as from relatives) to make what is a life-threatening decision. In my experience, however, when clinicians truthfully tell patients that they will die quickly without the transfusion, most consent. Some clinicians, steeped in the idea of patient autonomy, forget that informed consent includes informing the patient of all the relevant benefits and risks—including death.

Values in Emergency Medicine

Values describe the standards that individuals, institutions, professions, and societies use to judge human behavior. We learn values, usually at an early age, through indoctrination into the birth culture, from observing behavior, and through secular (including professional) and religious education. They are moral rules derived from ethical principles that promote those things we think of as good and minimize or avoid those things we think of as bad. Societal institutions incorporate and promulgate values, often attempting to retain old values even in a changing society.

In pluralistic societies, clinicians must be sensitive to alternative beliefs and traditions, since they treat people with multiple and differing value systems. Not only religious but also family, cultural, and other values contribute to patients' decisions about their medical care; without asking the patient, there is no way to know what decision they will make [2].

Although many people cannot answer the question "What are your values?", physicians can get concrete expressions of patients' uncoerced values by asking what they see as their goal of medical therapy and why they want specific interventions. In patients who are too young or who are deemed incompetent to express their values, physicians may need either to make general assumptions about what a normal person would want done or to rely on surrogate decision-makers [2].

Institutions, including healthcare facilities and professional organizations, have their own value systems. Healthcare facilities often have specific value-related missions. Religiously oriented or affiliated institutions may be the most obvious of these, but charitable, for-profit, and academic institutions also have specific role-related values. Professional organizations' values often appear in their ethical codes [3].

Clinicians also have their own ethical values, based on religious, philosophical, or professional convictions. While conscience clauses permit clinicians to "opt out" when they feel that they have a moral conflict with professionally, institutionally, or legally required actions, they are generally required to provide timely and adequate medical care for the patient—which may be particularly difficult to achieve in emergency medicine [3].

Virtues in Emergency Medicine

Virtues describe admirable personal behavior that Aristotle and other philosophers claim is derived from natural internal tendencies [4]. The virtuous person concept can be summed up with the ancient saying: "In a place where there are no men, strive to be a man!" [5]. Virtuous behavior stems from a sense of duty and the perception that it is the right thing to do, rather than from a desire to garner personal benefits. These ideal, morally praiseworthy character traits (e.g., showing kindness) are evident across many situations throughout the person's lifetime. Virtues that may be inherent in emergency medicine clinicians include courage, safety, impartiality, personal integrity, trustworthiness, and fidelity [1].

Courage allows one to fulfill an obligation despite reasonable personal risk. The courageous clinician also advocates for patients against incompetent practitioners and those who attempt to deny them care, autonomy, or confidentiality. *Safety* balances unreasoned courage. *Impartiality* prompts the emergency physician (EP) to provide unbiased, unprejudiced, and equitable treatment to all patients, without regard to their race, creeds, customs, habits, or lifestyle preferences. *Personal integrity* incorporates *trustworthiness*, which prompts clinicians to protect their sick and, often, vulnerable emergency patients' interests by exercising ethical principles. *Truth telling* (fidelity, honesty) prompts clinicians to provide patients with the known facts, but tempered with humility and sensitivity.

Bioethics, Religion, and Law

Religion Organized religions have long been recognized as the guardians of a society's values. Religious values have therefore been an important component of ethical deliberations in medicine, as elsewhere in society. Modern secular bioethics incorporates many religion-originated decision-making methods, arguments, and ideals [6]. Although various religions may appear to be dissimilar, most have as a basic tenet (no matter how it is stated) the Golden Rule: "Do unto others as you would have them do unto you." Religious values are important from two perspectives: the patient's in the exercise of autonomy and the practitioner's in placing limitations on what he or she can morally do. Given the overwhelming importance of patient autonomy in modern Western bioethics and law, however, a practitioner's religious convictions can only guide his or her actions. If their values differ, clinicians must follow the patient's wishes, as long as they are legal and practicable, and they do not violate medicine's basic ethical precepts.

Law Laws are rules of conduct established by legislatures, administrative agencies, courts, or other governing bodies. They often vary from locale to locale and are enforceable only in the jurisdiction where they prevail. Law and bioethics both provide rules of conduct to follow based on societal values. But, while good ethics often makes good law, good law does not necessarily make good ethics [6].

So, how does bioethics differ from law? The law, unlike bioethics, is relatively rigid and, particularly in the case of scientific and medical issues, can lag years or even decades behind modern developments. Societal values are incorporated both within the law and within ethical principles and decisions. By contrast, ethics is more inclusive within a culture, incorporating the broad values and beliefs of correct conduct. The primary differences between law and bioethics are shown in Table 64.2 [3].

Tat	ble	64	4.2	2	Re	lati	ions	hip	betwe	en 1	the	law	and	bioeth	nics
-----	-----	----	-----	---	----	------	------	-----	-------	------	-----	-----	-----	--------	------

Bioethics	Function	Law
✓	Case-based (casuistic)	1
1	Existed from ancient times	1
1	Changes over time	1
1	Strives for consistency	1
1	Incorporates societal values	1
1	Basis for healthcare policies	1
	Some unchangeable directives	✓
	Formal rules for process	1
	Adversarial	✓
1	Relies heavily on individual values	
1	Interpretable by medical personnel	
1	Ability to respond relatively rapidly to changing environment	

Reprinted from Iserson [36], with permission from Elsevier

Emergency physicians often look to the law for answers to thorny dilemmas. Yet, except for the rare cases of "black-letter law" wherein very specific actions are mandated, clinicians can best resolve these issues by turning to bioethical reasoning, using bioethics consultations, or applying previously developed institutional bioethics policies.

Modern bioethics developed because the law often has remained silent or inconsistent on matters vital to the biomedical community. The rapid increase in biotechnology, the failure of both the legal system and legislatures to deal with new and pressing issues, and the increasing liability crisis drove the medical community to seek answers to the difficult questions that practitioners have to work through on a daily basis [3].

Oaths/Codes

Medical ethics, or bioethics, differs from ethics in other fields just as medicine differs from other professions. This is because physicians treat ill people who are dependent on them and vulnerable to exploitation. For this reason, physicians have used ethical codes since ancient times to guide their behavior. Modern physicians who deliver critical hematologic and oncologic emergency medical services still rely on this guidance to help resolve dilemmas.

Many healthcare professional organizations, including most involved with emergency care, have developed their own values statements, which they often incorporate into their ethical codes. These codes (and the associated oaths) promote moral standards that their members presumably agree with and are expected to follow. The interpretation of those principles often evolves, albeit sometimes slowly, as the larger society changes. For example, although the American Medical Association's Code of Ethics was first published in 1847, it was not until 2001 that it stated that the physician's primary responsibility should be to their patient. While existing medical professional codes differ markedly (Table 64.3), all try to provide a "bottom line"—that is, a minimally acceptable course of action [2].

Some professional oaths and codes conflate bioethics and professional etiquette. However, these two areas differ markedly: professional etiquette deals with standards governing the relationships and interactions between practitioners, while bioethics is concerned with basic moral values and patient-centered issues [7]. Specifically, bioethics deals with relationships between providers and their patients, providers and society, and society and patients.

Review of Basic Ethical (Foundational) Theories

Foundational ethical theories embody grand philosophical ideas that attempt to coherently and systematically answer two

Principle or concept	Society for Academic Emergency Medicine	American College of Emergency Physicians	Emergency Medicine Residents' Association	American Medical Association	American Osteopathic Association (adopted by the American College of Osteopathic Emergency Physicians)	Hippocratic Oath
Protect patient confidentiality		X	Х	Х	X	Х
Maintain professional expertise	Х	Х	Х	Х	Х	Х
Committed to serve humanity	Х	Х	Х	Х	Х	
Patient welfare primary concern	Х	Х	Х	Х		Х
Considerate to patients, colleagues	Х	Х	Х	Х		Х
Respect human dignity	Х	Х	Х	Х	Х	Х
Safeguard public health	Х	Х	Х	Х		Х
Protect vulnerable populations	Х	Х	Х	Х	Х	Х
Advance professional ideals	Х	Х	Х	Х	Х	Х
Honesty		Х	Х	Х	Х	
Report incompetent, dishonest, impaired physicians		Х	Х	Х		
Moral sensitivity	Х	Х		Х		
Obtain necessary consultation				Х	Х	Х
Altruism in teaching	Х					
Fairness to students, colleagues				Х		Х
Obey, respect the law			Х	Х	Х	
Prudent resource use	Х	Х	Х			
Work to change laws for patient benefit			Х	Х		
Not abuse position or privileges, including sexual harassment	Х				Х	Х
Respect for students	Х					Х
Choose whom to serve except in emergencies				Х	Х	
Follow research guidelines	Х				Х	
No abortions						Х
No euthanasia						Х
Do not compromise clinical judgment for money					Х	
Universal access to				Х		
healthcare						
Preserve human life						Х

 Table 64.3
 Comparison of the ethical precepts contained in the Hippocratic Oath and five ethical codes used by emergency medicine professional organizations

American College of Emergency Physicians (ACEP). Code of Ethics for Emergency Physicians. Revised Jan 2017. www.acep.org/patient-care/ policy-statements/code-of-ethics-for-emergency-physicians/. Accessed 22 Sep 2019; Emergency Medicine Residents' Association (EMRA). Code of Ethics for Emergency Physicians. Revised Jun 2019; American Medical Association (AMA). AMA Code of Medical Ethics, Revised Jun 2001; American Osteopathic Association (AOA/AOCEP). The AOA's Code of Ethics. Updated 24 Jul 2016. https://osteopathic.org/about/leadership/aoa. governance-documents/code -of-ethics/. Accessed 22 Sep 2019; Hippocratic Oath. Miles [37] fundamental questions: What ought I do? How ought I live? Philosophers continue to elaborate or reconstruct fundamental ethical theories based on ancient ethical systems. Many were developed in India and China or within the Jewish, Christian, Islamic, and Buddhist religions. Clinicians generally have difficulty directly applying these theories to individual situations. Rather, they rely on "casuistry," a case-based application of bioethical values (described later in this chapter).

There are two main "foundational" theories of ethics: utilitarianism and deontology.

Utilitarianism, sometimes called consequentialism or teleology, is one of the more functional and commonly used ethical theories. Based on writings by John Stuart Mill and Jeremy Bentham, it focuses on getting good or valued results rather than using the right means to achieve those results. This theory promotes achieving outcomes that benefit the majority in the most impartial way possible. In its simplest form, this theory proposes achieving the greatest good (or the greatest sum of pleasure or the least amount of pain) for the greatest number of people. It is often advocated as the basis for broad social policies. Health planners often employ concepts of utility to develop more equitable health delivery systems. Such systems attempt to encourage and maximize the use of treatment that results in the most beneficial outcome for the least resource expenditure. Nevertheless, trying to define what is "good" or who comprises the affected community exposes the major problems with this theory [8].

Utilitarian principles apply to ED triage systems that regulate the resources given to each patient to maximize overall benefit. However, physicians should not use the utility concept as an excuse to deny an individual patient needed and available resources merely to add to society's greater good. In doing this, the physician would be abandoning the traditional healer's role and violating the bioethical principle of beneficence.

Deontology (rule-based ethics) is based on moral absolutes—something is either right or wrong. Adherents hold that certain unbreakable moral rules govern the most important aspects of our lives, even if following the rule leads to results that may not be "good." One example of a list of "unbreakable" rules is the Ten Commandments. The philosopher Immanuel Kant is often identified with this theory.

However, major problems can arise in applying rulebased ethics. The first is that moral rules may vary depending on one's culture or subculture. This can lead to great divisiveness over the interpretation of what might seem, at first glance, to be an obvious and straightforward rule. For example, does the common stricture "Do not kill" prohibit passive euthanasia (allowing death without intervening) or physicianassisted suicide (providing a patient with a lethal medication prescription)? The rigidity inherent in rule-based ethics causes difficulties when confronted with real-life situations. For some individuals, however, such a system provides necessary guidelines on how to conduct oneself in life. Other commonly cited ethical theories include:

Natural Law

This system, often attributed to Aristotle, suggests that man should live life according to his inherent human nature, in contrast to man-made or judicial law. Natural law is often associated with particular religious beliefs, especially Catholicism. The claim that the medical profession has an inherent morality mirrors natural law.

Virtue Theory

This theory asks what a "good person" would do in specific real-life situations. It stems from the writings of Aristotle, Plato, and Thomas Aquinas in which they discuss such timeless and cross-cultural character traits as courage, temperance, wisdom, justice, faith, and charity. The Society for Academic Emergency Medicine adopted a virtue-based Code of Conduct.

Mid-Level Ethical Principles

"Mid-level principles" that guide clinical practice and bioethical thought are derived from ethical theories, but are more specific and less abstract. Instead, these ethical principles are "action-guides," basically, role-specific duties that physicians owe to patients, consisting of various "moral rules" that comprise a society's values [9].

By melding medicine's goals with societal morality, law, religious values, and societal expectations for the profession, Beauchamp and Childress popularized the most commonly cited mid-level principles: autonomy, beneficence, nonmaleficence, and distributive justice. These four principles provide a handy medical ethics template and a practical, although often difficult to apply, checklist to use when considering the moral implications of specific cases [8, 9].

A question that naturally arises is whether ethical principles are universal. For individual clinicians, the bioethical principles they follow, and the values that stem from them, do not change because of geography. Clinicians practicing or teaching within cultures other than their own have a responsibility to continue applying their core ethical principles while being sensitive to the local population's values [10].

I will discuss autonomy in more depth (below), since it directly affects many decisions and ethical dilemmas that emergency clinicians face when caring for patients with hematologic-oncologic problems. These include whether a patient has the capacity to make his or her own decisions, who can act as surrogate decision-makers, and what is the role of advance directives. The other principles—and virtues—will be discussed in relationship to specific ethical dilemmas, such as demanding and refusing treatment, constraints imposed by healthcare systems, and patient/survivor notifications.

Decision-Making Capacity

Autonomy means, as Justice Cardozo said, "Every human being of adult years and sound mind has a right to determine what shall be done with his own body" [11]. Physician adoption of patient autonomy has been a major change from the millennia-old tradition of medical paternalism (or parentalism), that is, doing what the physician thinks is good for the patient regardless of what the patient desires. Grounded in the moral principle of respect for persons, autonomy recognizes the right of adults with decision-making capacity to accept or reject recommended healthcare interventions, even to the extent of refusing potentially lifesaving care. Physicians have a concomitant duty to respect their choices. Over the past several decades, autonomy has become the predominant value in US medicine and society, although paternalism is still the prevailing attitude in most of the world.

One important, and often misunderstood, aspect of autonomy is that individuals who retain decision-making capacity can voluntarily and verbally assign decision-making authority to other people (e.g., family) for a specific decision or time period, such as when they are in the emergency department. Since patients may exercise their autonomy only if they have decision-making capacity, emergency clinicians must be able to determine this at the bedside so that if necessary, surrogate decision-makers may become involved.

While autonomy has become ingrained in US medical professionals, clinicians need to be sensitive to communitarianism. which is a counterbalance to autonomy. Communitarianism considers the larger picture of the patient's life, including his or her family and community, when puzzling through a bioethics case or developing public policy. This principle generally holds that the community's welfare outweighs an individual's rights or good and thus requires that deliberations involve communal (e.g., family, elders) discussions [8]. Many cultures rely on communitarian deliberations when making medical choices and use this pattern for public policy decisions. When making bedside ethical decisions, physicians should determine, whenever possible, not only their patient's individual values but also whether the patient subscribes to an individualistic or a communitarian ethic [6].

Evaluating Decision-Making Capacity

Many ethical dilemmas in emergency medical care revolve around ascertaining a patient's decision-making capacity. In clinical settings, the question of decisional capacity is most often linked with consent to (or, more often, refusal of) a medical procedure.

Capacity refers to a patient's decision-making ability that, in the ED, EPs determine at the bedside rather than by the Table 64.4 Components of decision-making capacity

- 1. Knowledge of the options
- 2. Awareness of consequences of each option
- 3. Appreciation of personal costs and benefits of options in relation to relatively stable values and preferences

From Buchanan [38]. \bigcirc 1995 by Galen Press, Ltd. All rights reserved. Used with permission of Galen Press, Ltd., Tucson, AZ

courts, a psychiatrist, or a lawyer. ("Competence" is a legal term and can only be determined by the court.) Decisional capacity is always related to the type of decision involved, although it is unclear whether it should be based on the potential seriousness or irreversibility of the outcome of a patient's decision (e.g., refusing lifesaving intubation) or on the complexity of the information needed to make the decision (e.g., whether to enter an experimental cancer treatment protocol). In current practice, most clinicians and ethicists use the seriousness or irreversibility of the outcome as the key to determining decisional capacity.

To have adequate decision-making capacity in any circumstance, an individual must understand (a) the options, (b) the consequences of acting on the various options, and (c) the personal costs and benefits of these consequences related to a relatively stable framework of personal values and priorities (Table 64.4) [12]. Assessing this last criterion can be especially difficult when clinicians have poor verbal skills in the patient's language. An easier, albeit incomplete, method of assessing this criterion is to ask the patient "why" a particular decision was made. This often provides an approximation of the last (and most important) criterion for assessing decisional capacity.

Disagreement with the physician's recommendation is not in itself grounds for determining that the patient is incapable of making his own decisions. In fact, even refusal of lifesaving medical care may not prove that the person is incapable of making valid decisions if he or she makes it on the basis of firmly held religious beliefs, as when a Jehovah's Witness patient refuses a blood transfusion.

Patient Consent

If a patient has decision-making capacity, a clinician who respects a patient's autonomy must get the patient's consent for any intervention. The consent need not be associated with a formal document, although an appropriate level of explanation is always required.

There are three general types of consent: presumed, implied, and informed. Presumed consent, sometimes called emergency consent, covers the necessary lifesaving procedures that reasonable people would usually wish to have performed on them. *Presumed consent* conjoins a patient's "best interest" with physician beneficence. Stopping hemorrhage and securing an airway in an unconscious, unknown patient are common examples of procedures performed under this type of consent. *Implied consent* occurs when a patient with decision-making capacity simply cooperates with a procedure, such as holding out their arm to give blood or to allow placement of an intravenous line. Indeed, this is the most common type of consent in medical practice [13].

Informed consent occurs when a patient who retains decision-making capacity is given all the pertinent facts regarding a particular procedure's risks and benefits, understands them, and voluntarily agrees to undergo the procedure. The requirement for informed consent varies in practice and law from area to area and even among practitioners and institutions in the same area. If a patient lacks decision-making capacity, get a surrogate decision-maker involved.

Advance Directives and Surrogate Decision-Makers

Advance directives loosely include durable powers of attorney for healthcare, living wills, prehospital advance directives [14], and similar documents initiated or approved by physicians, such as prehospital DNAR, inpatient DNAR forms, and Physician Orders for Life-Sustaining Treatment (POLST). They do not, however, include nonstandard and indecipherable directives [15–17]. The standard and generally recognized documents often express the patient's autonomous wishes about the treatment he or she will receive. However, they only go into effect if the patient lacks decision-making capacity. Otherwise, ask the patient what he or she wants done.

When patients do not have the capacity to make medical decisions for themselves, someone must make the decision for them. Four major classes of decision-makers have been proposed, and actually used, in these situations: family, bio-ethics committees, physicians, and courts.

Traditionally, and usually in practice, the family, especially the spouse, makes medical decisions when a patient does not have decision-making capacity. A typical prioritization list of those empowered to act as surrogate decision-makers is often stipulated in state statutes, similar to Arizona's landmark law (Table 64.5) or in a hospital's policy. When no surrogates exist, all potential surrogates refuse to act in that capacity, or an irresolvable conflict exists between surrogates at the same level (such as siblings), the court will intervene.

Surrogates make decisions in one of two ways. The first is *substituted judgment*, which is used when the surrogate is not certain what the patient would want done in a particular situation. Substituted judgment attempts to determine and act in accordance with the patient's values based on the patient's prior statements and behavior, without advance directives or other explicit direction. This is the most worrisome type of surrogate decision-making, because it is based on the most ambiguous grounds. The second way, or *best interest* standard, is used when the patient has never had ade-

 Table 64.5
 Statutory surrogate decision-maker list: an example

Arizona Revised Statute: Living Wills and Health Care Directives Act, Title 36, Chap 32. 1992. Revised 2005

- 1. The patient's *spouse*, unless the patient and spouse are legally separated
- 2. An *adult child* of the patient. If the patient has more than one adult child, the healthcare provider shall seek the consent of a majority of the adult children who are reasonably available for consultation
- 3. A *parent* of the patient
- 4. If the patient is unmarried, the patient's *domestic partner* if no other person has assumed any financial responsibility for the patient
- 5. A brother or sister of the patient
- 6. A *close friend* of the patient. For the purposes of this paragraph, "close friend" means an adult who has exhibited special care and concern for the patient, who is familiar with the patient's healthcare views and desires, and who is willing and able to become involved in the patient's healthcare and to act in the patient's best interest
- 7. If the healthcare provider cannot locate any of the people listed [above], the patient's *attending physician* may make healthcare treatment decisions for the patient after the physician consults with and obtains the recommendations of an *institutional ethics committee*. If this is not possible, the physician may make these decisions after consulting with a second physician who concurs with the physician's decision. For the purposes of this subsection, "institutional ethics committee" means a standing committee of a licensed healthcare institution appointed or elected to render advice concerning ethical issues involving medical treatment

quate decision-making capacity, and the surrogate must simply act in the patient's best interest. Unless there is already a court-appointed guardian, these cases often end up being resolved in a courtroom.

Children pose a special situation. Individuals less than the age of majority (and unemancipated) are usually deemed incapable of making medical decisions for themselves, although clinicians normally explain the situation to the child and ask for his or her assent. In most cases, the same rules for decision-making capacity that apply to adults also apply to children. The more serious the consequences, the more important it is that the child understands the options and consequences and can articulate the values involved in making their decision. Especially in cases involving religiously or culturally based refusal of potentially lifesaving treatment or when the parents disagree, the court or child protective services may intervene on the child's behalf.

Methods of Applying Bioethics Principles

To apply bioethical principles to a clinical situation, one first must recognize that a bioethical dilemma exists, which is not always an easy task. Once identified, addressing the problem brings its own challenges. Clinicians adhere not only to basic bioethical principles but also, at least tacitly, to a number of professional, religious, and social organizations' ethical oaths, codes, and statements. This complexity can produce a confusing array of potentially conflicting bioethical imperatives.

When dealing with bioethics cases, clinicians need to use ethical reasoning, which includes the application of foundational theories, mid-level principles, and case-based reasoning. This helps us systematically identify elements within moral problems that we otherwise might overlook.

Casuistry, or case-based ethics, attempts to define problems and correct courses of action based on the intricacies of a particular case. It puts an emphasis on what Aristotle called *phronesis*, or "practical wisdom," and is the basis for the emergency rapid decision-making model, described below. To use this method, examine each case for its similarities and differences with select previous cases paradigms, for which you have determined a suitable course of action. Where the present case is similar enough to the paradigm, use the same course of action. When significant differences exist, clinicians must apply the broader mid-level principles derived from rulebased, utilitarian, and other ethical systems, usually giving the most weight to patients' autonomy and values.

In practice it can be difficult to identify and extract the most appropriate and useful principles to apply to a particular case. Some principles may appear too vague, or perhaps several conflicting principles appear to apply to a given case. The key is to prepare for bioethical problems as one would for critical medical events, by reading about, reflecting on, and discussing how to approach these issues. This leads not only to increased personal preparation but also to more general policies that provide guidance for dealing with difficult bioethical issues [2].

Prioritizing Conflicting Principles: The Bioethical Dilemma

Applying bioethical principles can be confusing. When two or more seemingly equivalent principles or values appear to compel the clinician to act in different ways, a bioethical dilemma exists. This situation is often described as being "damned if you do and damned if you don't," where any potential action appears, on first reflection, to be an option between two seemingly equivalent "goods" or "evils." In bioethics, although there may be disagreements regarding the optimal course of action using a specific set of values, there is often general agreement as to what constitutes ethically wrong actions. While we theoretically have a duty to uphold each bioethical principle, none routinely "trumps" another.

Working through bioethical dilemmas generally requires a case-based approach. The key is to use paradigm and analogy (the first step in the rapid decision-making model, described below). Thus, when faced with a troubling case, first identify relevant mid-level principles and alternative courses of action. Then, compare it to similar but much clearer paradigms, that is, cases having resolutions with which virtually any "reasonable person" will agree. Identifying such cases may be difficult; it takes experience and a significant knowledge base. Using bioethics committees and bioethical or legal case databases may help.

Application to Emergency Medicine: The Rapid Decision-Making Model [18, 19]

When faced with bioethical dilemmas, emergency clinicians often must make ethical decisions with little time for reflection or consultation. Ethical problems, like clinical problems, require action for resolution. For that reason, a rapid decision-making model was developed, based on accepted bioethical theories and techniques (Fig. 64.1). It provides guidance for emergency medicine practitioners who are under severe time pressures and wish to make ethically appropriate decisions [18, 19].

When using this approach, the clinician must first ask: "Is this an instance of a type of ethical problem for which I have already worked out a rule?" Or, at least, is it similar enough to such cases that the rule could be reasonably extended to cover it? In other words, if there had been time in the past to think coolly about the issues, read about them, discuss them with colleagues, and develop some rough guidelines, could they be used in this case? Just as with the indications for any clinical emergency procedure, EPs should be prepared with a course of action for at least the most common ethical dilemmas likely to occur in the ED. If the case in question does fit under one of those guidelines arrived at through critical reflection, and there is not time to further analyze the situation, then the most reasonable step would be to follow that rule-if it is still appropriate. In ethics, this step follows from casuistry or case-based reasoning.

If the case does not fit under any previously generated ethical rule, the practitioner should consider if there is an option that will buy time for deliberation. If there is such an option, and it does not involve unacceptable patient risks, then it would be the reasonable course to take. Using a delaying tactic may afford time to consult with other professionals, the bioethics committee, and the family.

If there is no acceptable delaying tactic, the clinician should weigh what she considers the best option using a set of three tests, drawn from three different philosophical theories, to help make a decision:

Impartiality Test "Would you be willing to have this action performed if you were in the other person's (the patient's) place?" A version of the Golden Rule, it helps correct one obvious source of moral error—partiality or self-interested bias. **Fig. 64.1** Rapid decisionmaking model. (From Iserson [39], Figure 2, pg 45. © 1995 by Galen Press, Ltd. All rights reserved. Used with permission of Galen Press, Ltd., Tucson, AZ)



Universalizability Test "Would you be comfortable if all clinicians with your background and in the same circumstances act as you are proposing to do?" This generalizes the action and asks whether developing a universal rule for the contemplated behavior is reasonable—an application of Kant's categorical imperative. This helps eliminate not only bias and partiality but also short-sightedness.

Interpersonal Justifiability Test "Can you give reasons that you would be willing to state publicly? Will peers, superiors, or the public be satisfied with the answers?" This uses a theory of consensus values as a final screen.

When ethical situations arise in cases for which no time exists for further deliberation, it is probably best to go ahead and act on the previously determined ethical rule or take the course of action for which the clinician was able to answer all three tests in the affirmative with some degree of confidence. Once the crisis has subsided, clinicians can hone their ethical decision-making abilities by reviewing the decision with colleagues and bioethicists. As an example:

Case Study 2

An elderly man arrives at the ED with a diminished state of consciousness and a diminished respiratory rate. The EMS personnel who are bagging him say that they were told that he has "cancer," but had no additional information. No one accompanied him. Upon removing his shirt, they find "Do Not Resuscitate" tattooed over his left chest. What should the EP do?

The first step, as with the clinical situation, is for the EP to ask herself whether she has previously encountered—personally, through reading, or at conferences—this type of case. A number of these cases describing nonstandard advance directives, including tattoos, bracelets, and neck-laces, have been described in the medical literature [15–17]. If she is aware of this, do they give her an ethically justifiable plan of action? For several reasons, nonstandard tattoos cannot be honored. The reasons include the unreliability of knowing whether the patient really wouldn't have wanted resuscitation in the current situation and whether the patient's situation since the tattoo was placed has changed. If she knows this and agrees, she should resuscitate.

If she does not believe that this case fits under any previously generated ethical rule, she should consider buying time for deliberation. Naloxone, artificial ventilation, and hydration would allow time to search for the patient's identity, medical records, and a possible surrogate if still needed. As described elsewhere in this chapter, initiating these interventions does not mean that they cannot be withdrawn if additional information makes that choice reasonable.

If the EP does not feel that this is reasonable, she can select what she considers the best ethically appropriate option and use the three tests to evaluate that choice. (Impartiality Test) If we assume that she chooses not to resuscitate, she should ask herself (Impartiality Test) whether she would want to have clinicians let her die (the patient is being kept alive with bagging) if she was the patient with an uncertain diagnosis, history, and wishes. That is unlikely. If she says no, then she must go to another choice, such as to resuscitate. (Impartiality Test) Would she approve of that choice for herself? (Universalizability Test) Would she approve all other EPs resuscitating in similar situations? (Interpersonal Justifiability Test) Would she be willing to defend this action to her peers, superiors, or the public? If all three answers are in the affirmative, she should proceed with the assurance that she will be within the spectrum of ethically appropriate actions.

Bioethics Committees and Consultants

Another resource for complicated ethical dilemmas is to use your institution's bioethics committee. Most US hospitals now have multidisciplinary committees or bioethics consultants to help resolve bioethical dilemmas. Bioethics committees and consultants have four roles: (a) education, (b) policy development (proactive ethics), (c) retrospective case review, and (d) concurrent case review (ongoing clinical cases in which they often mediate between dissenting parties) [20]. Some experienced committees and consultants also perform "stat" consultations that can assist in emergency department cases.

Other Principles and Virtues

Other mid-level bioethics principles and virtues often guide clinician behavior. They also may conflict with the principle of autonomy or with each other, posing a bioethical dilemma. In their practice, emergency clinicians commonly use the principles of beneficence and nonmaleficence, as well as the virtue of truth telling. When developing policy, they often use the principle of distributive justice. Therefore, it is instructive to examine how these principles relate to specific clinical scenarios with ED patients, including those with hematologic or oncologic illnesses.

Beneficence is the principle of doing good or producing benefits. This principle is one of the medical profession's universal tenets. Society's view of physicians as altruistic reflects the profession's long history of beneficence. In addition, all medical students are taught the basic tenet of nonmaleficence: *primum non nocere* (first, do no harm). This stems from recognizing that physicians can harm, as well as help, their patients.

Clinicians use the principle of distributive justice to develop policies, including triage protocols, affecting patient groups and healthcare systems. Truth telling is the virtue that guides clinicians in what and how they communicate with patients and families, rather than the decisions they must make.

Beneficence

Clinicians enter the healthcare field to help others—to be beneficent or to do good. While ED interventions for hematology-oncologic patients will not provide a long-term solution, they often relieve symptoms or provide time to begin more definitive treatments. However, when opportunities to clearly benefit a patient present themselves, clinicians feel intense anguish when a patient or surrogate decisionmaker refuses the interventions. This sets up a struggle between patient autonomy and physician beneficence. Probably the most common ethical dilemma in modern US medical practice, it exemplifies physician paternalism, that is, the desire to do what he or she thinks is best for the patient no matter what the patient (or surrogate decision-maker) wants.

Yet, when made by patients with decision-making capacity, clinicians should respect these refusals. That does not mean that the clinician should not clearly explain the options, potential outcomes, and costs involved. If the patient holds firm to the decision, the clinician must follow the patient's wishes, even if they conflict with his or her own values. This is the most difficult part of adhering to patient autonomy.

The only exceptions to this are when a surrogate makes a decision that the clinician believes is contrary to the patient's expressed wishes or is masking (possibly illegal) ulterior motives, or when a child is involved. In any of these situations, obtain legal assistance immediately. In the case of a child, including religion-based refusals of treatment, most courts will order clinicians to institute therapy if any reasonable chance of benefit exists.

Beneficence: Withholding and Withdrawing Treatment

As noted above, resuscitating patients who present to the ED with unknown illnesses and injuries is both ethically appropriate and virtuous behavior. A common fear, and unfortunate misunderstanding, is that once treatment is initiated, it cannot be withdrawn. Actually, there is a much higher ethical and legal bar to withholding treatment in uncertain cases than there is to withdrawing treatment once complete information is known [21, 22].

Withholding Treatment

Not infrequently, a patient is brought into the ED in extremis, unable to interact with clinicians, and without any history or direction about care. For example, the patient may be in cardiorespiratory failure or the patient may have metastatic cancer and now is suffering from hypercalcemia, a frequent terminal event. While some have advocated that allowing the patient with hypercalcemia to have a "good death" may be humane and medically appropriate [23], EPs do not have this option. Without knowing the patient, the disease prognosis, or any prior wishes, they are obligated to intervene to preserve life. This obligation is based on the principles of beneficence and nonmaleficence, which are societal values placed on EPs. Our society sees the entire emergency medical care system as being the caregivers of last resort. Arbitrary decisions to do less than everything reasonable to preserve a life signal a lapse in this entrusted function. Unknown and unknowing patients deserve the presumption of life.

Withdrawing Treatment

Contrary to popular myth, if the EP (or inpatient physician) later learns that, given the patient's condition or wishes, lifesaving interventions such as ventilation and vasopressors are not appropriate, it is both ethical and legal to withdraw them. This follows the dicta to use only beneficial interventions and to preserve a patient's autonomous wishes. Morally, withdrawing treatment is identical to initially withholding it. That is, withdrawing an IV drip or stopping a ventilator is equivalent to withholding the next drop of medication or the next ventilation. The problems that generally arise with withdrawal under these circumstances are emotional, not ethical [21].

Even though treatment has been withdrawn, clinicians must continue to provide analgesia and any other appropriate care. Healthcare professionals never cease providing care.

Beneficence Versus Patient Autonomy: Refusing Lifesaving Treatment

As the relatively common case described in Case 1 demonstrates, the ethical dilemma is produced by the tension between the physician's motivation of beneficence and the patient's (or surrogate's) desire to determine which treatments to authorize based on his or her values. In this case, the decision is religiously based.

Beneficence Versus Patient Autonomy: Refusing Analgesia

Physicians are expected to follow the medical maxim "cure sometimes, relieve often, comfort always" [24]. In some cases, patients or their surrogates may refuse analgesics to relieve acute pain. This may be due to misguided concepts of drug abuse and addiction or to a fear that taking analgesics will hasten death. Rarely, refusal may stem from religious or cultural values.

The final decision may come down to a balance between autonomy and beneficence. While there may be unique instances when analgesics should be withheld, at least in the short term (e.g., so that the patient can be awake when relatives arrive), beneficence generally outweighs any countervailing argument and the patient should receive analgesia.

Nonmaleficence

The principle of nonmaleficence includes not doing intentional harm to patients, preventing harm, and removing harmful conditions. Nonmaleficence is the profession's protective shield for patients. The following two situations demonstrate how this may not only conflict with other principles, such as autonomy, but also how it forms the basis for the rules regarding clinical research.

Nonmaleficence: Demands to "Do Everything"

No one gets every possible medical intervention. Yet, ED clinicians commonly hear surrogates demand that they "do everything," even for terminally ill hematology-oncologic patients for whom further intervention will not change the disease course and may prolong an unpleasant dying process. This request, often coming from distraught and guilt-stricken relatives, poses difficult ethical dilemmas for clinicians. While patient autonomy plays a key role in any decision, surrogates may be unaware that clinicians' interventions must not harm the patient without providing them with a countervailing benefit (nonmaleficence).

The "do everything" request usually presents as one of three scenarios: where a patient knowingly requests intervention, where a patient asked for intervention via an advance directive, or where surrogates ask for the intervention.

The first situation occurs when a patient with decisional capacity who is informed of the options selects a probably non-beneficial and definitely painful course of therapy. In the ED, that may mean intubating and ventilating a terminal cancer patient in severe pain. These decisions fall under the question of patient autonomy, and even if the physician thinks she would not make the same decision herself, she should help the patient implement this choice.

The second scenario occurs when a patient has left instructions via an advance directive to "do everything." This directive carries much less weight than the patient's actual informed decision, described above, because the exact situation with which the medical team is presented could not have been anticipated. Nevertheless, clinicians should make all reasonable efforts to comply with the patient's wishes. The following case illustrates just such a dilemma for the EP.

Case Study 3

A patient with decision-making capacity completed a new Durable Power of Attorney (naming her husband as surrogate) and a Living Will upon entering the hospital three weeks earlier for the treatment of metastatic breast cancer. Both documents indicate that she wants everything possible done to save her life. She now comes to the ED with markedly diminished consciousness, bradycardia, dehydration, but a normal respiratory rate. Her husband accompanies her. Suspecting hypercalcemia, the EP prepares to initiate normal saline hydration and appropriate laboratory studies when he receives a call from the patient's oncologist who tells him to do no studies or interventions. When informed about the advance directives and the potentially correctible condition, he says that he and another physician agree that nothing more should be done for this patient, despite what any documents or surrogates say.

Although some institutional policies now suggest that two physicians can initiate a Do Not Attempt Resuscitation order that countermands a patient or surrogate's wishes, this violates both ethical and legal norms. The EP's role is to provide potentially beneficial treatment (and care) whenever practicable. Admitting physicians may later wish to do otherwise, but that is between them, the family, the bioethics committee, and, ultimately, the legal system.

The third situation occurs when families of a terminally ill patient demand non-beneficial care for their relative. Emergency physicians are usually reluctant to provide this, since it only prolongs the predictable dying process. On the other hand, to be beneficent, clinicians frequently admit endstage cancer patients if they come for pain relief that cannot be provided at home, to temporarily relieve a family of the stress of caring for the patient (respite care), as an interlude to get a patient into a hospice or nursing facility, or who are in the terminal stage of the disease presaging death. However, interventions which simply prolong dying usually violate the ethical principle of nonmaleficence.

Legally, the representative for a patient lacking decisional capacity can make any informed decision that the patient could make about healthcare. After explaining the options and that the interventions will not be beneficial, physicians should abide by these surrogates' requests, even if they seem unreasonable. Note, however, that a physician is never required to offer any treatment through a surrogate that they would not offer directly to a patient, such as cardiopulmonary resuscitation in an imminently dying metastatic cancer patient. This is a struggle between autonomy and nonmaleficence, and the medical team's responsibility is to follow the legal surrogate's instructions to the extent that they would follow a patient's instructions. The assumption is that in most cases, the patient believed that this individual would best represent his or her wishes. When clinicians question whether the agent is acting in the patient's best interest, they can ask a court intervene.

Nonmaleficence and Autonomy: Research Protocols

The horrors inflicted under the guise of scientific research during World War II led to the Nuremburg Code and subsequently the Helsinki Declaration, enumerating basic ethical principles for research studies [25]. Based on autonomy, the respect for persons as individuals, these research principles arose from the desire to no longer harm research subjects, as had been done both during WWII and subsequently in the civilian sector.

Research is vital to medicine. In the past, most medical care, including that in emergency medicine, has relied on experience that was unsupported by investigation, so-called nonvalidated practice. Recently, however, clinicians have begun to use evidence-based medicine, which requires research. Over the past three decades, research done within emergency medicine and that done elsewhere but applied to emergency medical practice has improved the elegance of patient encounters, significantly benefiting ED patients. In hematology-oncologic, research has driven diagnostic and treatment breakthroughs, and EPs can often assist in these projects.

Yet some aspects of clinical research and research oversight fall short of meeting the ethical standards of safety and patient benefit. Overall, emergency medicine research has been and continues to be a moral endeavor. Even more important than the institutional safeguards, such as the institutional review boards (IRBs), is the individual researcher's moral compass, which must serve to protect the subjectpatients of clinical research. Perhaps the greatest moral lapse has been the lack of attention to key populations, such as women and children, within emergency medicine research, with the result that patients most needing acute intervention are the ones who suffer [26].

Funding availability, both from private industry and from government agencies, still drives research agendas. This raises questions about clinical researchers' fiduciary responsibility to their subject-patients.

Finally, the moral responsibility to ensure that any research protocol and its execution are ethical extends to the journals in which the research is published [27]. While emergency medicine has an excellent record of ethical research, a large percentage of human research studies published in the major EM journals fail to mention either IRB review or informed consent [26, 28].

System Constraints: Distributive Justice and Confidentiality

Distributive or comparative justice suggests that comparable individuals and groups should share similarly in the society's benefits and burdens. In contrast to the judicial system's retributive and compensatory justice, this basic bioethical principle does not apply to individual practitioners for ad hoc use in limiting healthcare resources for individual patients [29]. Rather, it is meant to be used at the policy-making level to allocate limited healthcare resources.

For example, triage decisions conform to this principle when they are applied uniformly and impartially to all patients [30]. Other typical issues in emergency medicine for which distributive justice plays a part in designing policies and protocols include admission prioritization; how to work with patients who cannot pay for treatment; have the "wrong" insurance, or belong to the "wrong" medical system or group for the particular hospital or clinic; intentional or unintentional release of patient information; and how to work with patients who are undocumented aliens.

Other principles have also had long-standing importance to medical practice, one of the most important being confidentiality, that is, the nondisclosure of patient information. Based on a respect for persons (as is autonomy), patient confidentiality has been a cornerstone principle of the medical profession since antiquity. The Hippocratic Oath, for example, states, "Whatever, in the course of my practice, I may see or hear (even when not invited), whatever I may happen to obtain knowledge of, if it be not proper to repeat it, I will keep sacred and secret within my own breast." Confidentiality presumes that, unless they first obtain the patient's permission, physicians will not reveal to any other person or institution what patients tell them during the medical encounter. Various US federal and state laws have both emphasized (e.g., Health Insurance Portability and Accountability Act of 1996, HIPAA) and carved out exceptions (mandatory reporting) to this stricture. With the advent of minimally secure electronic medical records, the ability to maintain patient confidentiality has become even more difficult.

Note that privacy, often confused with but related to confidentiality, is a patient's right to sufficient physical and auditory isolation such that he or she cannot be seen or heard by others during interactions with medical personnel.

Truth Telling (Fidelity)

Truth telling remains a somewhat controversial virtue within the medical community. While many champion absolute honesty to the patient, honesty must be tempered with sensitivity and compassion; it should not equate to brutality. In recent years, poor role models, a lack of training in interpersonal interactions, and bad experiences may have diminished the perception of truth telling as a physician virtue. There are multiple tales of the champions of absolute fidelity who, nevertheless, were appalled by their own physician's lack of sensitivity when relating unfavorable medical news to them [31].

The degree to which physicians fail to disclose the truth varies with the circumstances. When failure to disclose the truth will do physical harm to the patient, such as in the infamous Tuskegee experiments on patients known to have syphilis, it is not only immoral but also probably illegal to withhold the information. Likewise, if failure to disclose information is strictly for the physician's benefit, such as telling a patient who calls in the middle of the night to "take two aspirins and call in the morning," although there is a strong suspicion of serious disease, there are serious ethical and legal deficits in the clinician's behavior. The issues become somewhat murkier when truth telling involves a third party, such as a sex partner who the patient has exposed to an infectious disease [32].

The following cases demonstrate two scenarios involving this principle that commonly occur with ED hematologyoncologic patients. The first deals with relating a probable diagnosis to a woman in a strong communitarian culture. The second deals with death notification, emphasizing the need for strong communication skills and sensitivity.

Truth Telling and Communitarianism: Diagnosis Notification

Case Study 4

A 54-year-old Hispanic woman comes to the ED with her family because of a persistent cough and poor health for at least the past several weeks. Before the patient can be examined or any tests performed, the patient's husband intercepts the EP and tells him that if the patient has a life-threatening disease, she is not to be told because "she doesn't want to know." The adult children agree. The evaluation shows that the woman has a hard new breast lump, honeycomb lesions, and multiple pulmonary nodules consistent with cancer. The physician has a policy to tell the truth to all his patients but believes that the family might be accurate in their assessment.

Many patients come from cultures that embrace communitarianism, rather than autonomy. Communitarianism stresses the interactions between group members, which may be just the family, but may also include elders, religious figures, or the entire tribe, group, or community. In this case, the family implied that the patient was part of such a culture.

Doing good in these cases often means respecting the patient's personal or cultural desire not to be explicitly informed about a serious disease. This is the norm for many Asians (particularly Japanese), Hispanics, and Native Americans. The enormity of this information (and slight possibility of error in this case), coupled with the minimal physician-patient relationship established in the ED, might also suggest that, at least at this stage, stating the presumed "diagnosis" could be avoided.

The question for the physician is, how much does the patient want to know? The best way to find out is to ask her both what she wants to know and, if she does not want to know anything, with whom does she want the physician to speak. If she wants the information, the physician is obligated to gently tell her what he knows about her illness, including the next steps in the diagnostic process. If she designates someone else to receive this information, this fully complies with the patient autonomy principle and should be followed.

Truth Telling: Survivor Notification

Nowhere in emergency medicine is truth telling with sensitivity more important than when the clinician must deliver the news of a death, which is often an emotional blow, precipitating life crises, and forever altering the survivors' world. Emergency physicians must repeatedly do death notification as part of their daily work.

Case Study 5

A 63-year-old man who had been in remission with lung cancer presents to the ED in cardiac arrest. Bystanders had told the EMS personnel that he had been at the store when he suddenly became dyspneic and collapsed. They administered ACLS for 20 minutes without success. Without vital signs or cardiac activity, you pronounce him dead with a probable diagnosis of pulmonary embolus. You are notified that his wife and an adult child have arrived and were waiting in a side room. They do not know he is dead; it was unexpected.

Kind but direct notification has been shown to be what most US and Canadian families desire. Other cultures prefer more indirect notifications. Come in, sit (or kneel) so that you are at eye level with the key recipient—in this case, the wife. Be sure that you or someone on your team has definitively identified the people as the decedent's family. Then say, "I'm sorry, but I have very bad news. Despite doing everything we (includes paramedics) could, your husband died." Truth telling and survivor understanding require that you use a "D" word, no matter how difficult that is. Otherwise, the message you relay may not be clearly understood. "Pass on," "No longer with us," "Gone to another place," and other euphemisms will not work. Take a deep breath and say "died" [33].

The wife then asks two things: "What did he die from?" and "Can you call his brother in another state and tell him?" The first answer is generally straightforward. No matter what your clinical suspicion is, truthfully tell her that you cannot be sure and that finding out will require an autopsy. (In some situations, a medicolegal autopsy might be automatic. If not, you can encourage the family to ask for one.) Telling the brother also requires truth telling. Gently inform a survivor who is distant from the ED about the death. Don't say that he or she is "critically ill." That way they won't make a heroic, sometimes dangerous, attempt to get to the hospital "in time" [34]. Excellent communication skills represent the basis for correctly delivering tragic news to survivors. Directness, truth, consistency, and clarity are the key factors in delivering information about a sudden, unexpected death—and complying with the virtue of fidelity. Perceptive survivors can easily tell which notifiers care and which are only "going through the motions" [33, 34].

Poor clinician-patient communication disappoints both the patients and clinicians. Often, this failure is due to clinicians:

- Using highly technical language
- Not showing appropriate concern for problems voiced by patients
- Not pausing sufficiently to listen
- Not verifying that the listener has gotten the information presented
- Using a generally impersonal approach to the interaction, including their manner of speech [35]

Delivering the news about sudden unexpected death provokes strong emotions in both the notifier and survivors. Communication is improved if the notifier acknowledges those emotions, being prepared to vocalize and demonstrate their sadness and to recognize and acknowledge it in the survivors. Using the voice to communicate does not always mean talking. In some instances, para-verbal behavior is what is called for. These sounds, such as mmmmm, ahhhh, or mhmmm, are often sufficient to show that a person is listening and understands, particularly if they are accompanied by appropriate nonverbal cues, such as nodding the head.

It often takes imagination to put oneself in the position of a grieving survivor, especially when wide cultural or age differences exist. Even if you cannot learn to empathize with survivors, you can learn to behave appropriately, speak correctly, and assist them in their time of grief. Imagination, studying people, advance planning, or taking the lead from experienced mentors is the only way to successfully perform this necessary, but tragic task [33, 34].

Future Bioethics in EM and the Role for Emergency Physicians

No patient can receive every possible diagnostic or therapeutic intervention. The clinician's traditional role has been to determine which resources would most likely benefit particular patients. As these resources become more costly and, in some cases less available, insurers, healthcare institutions, and families have become part of this decision-making process. In the future, bioethics discussions will become increasingly vital to protect patients and the integrity of the healthcare system. The bioethics field includes teachers, writers, and clinicians. Many individuals in the field are involved in more than one of these areas. People enter the field with a variety of backgrounds, the most common being medicine (e.g., physicians, nurses, and physician assistants), legal, religious, and philosophy. Many, if not most, individuals have additional training before embarking on a leadership role within bioethics. This may include fellowships, serving on an institutional bioethics committee, or working with an experienced mentor.

Those in clinical ethics need a grounding in the "Mid-Level Principles," as described in this chapter and elsewhere. In addition, especially for a leadership role, they should have a knowledge of group dynamics, negotiation, compassionate listening, and writing. All come into play in working with patients, families, other surrogates, treating clinicians, and healthcare institutions. Very few clinicians are reimbursed for their time doing clinical ethics.

Emergency physicians, partially due to their training and type of practice, have unique qualifications for clinical bioethics. Aside from their ability to negotiate (e.g., for admissions, consultations, special testing), they work with and are usually sensitive to the differences among multiple socioeconomic and cultural groups. Similarly, they work with and are generally familiar with a wide variety of healthcare professionals, institutions, and systems. All these elements often arise in the course of bioethical consultations, policy making, and clinical education.

References

- American College of Emergency Physicians Ethics Committee. Code of ethics for emergency physicians. Dallas, TX: ACEP; 1997.
- Iserson KV. Principles of medical ethics. In: Marco C, Schears R, editors. Clinical ethics in emergency medicine. Cambridge: Cambridge University Press; 2015.
- Iserson KV, Heine C. Bioethics. In: Marx JA, Hockberger RS, Walls RM, editors. Rosen's emergency medicine: concepts and clinical practice. 7th ed. Philadelphia, PA: Mosby; 2013. p. e33–46.
- Larkin GLL, Iserson KV, Kassutto Z, Freas G, Delaney K, Krimm J, et al. Virtue in emergency medicine. Acad Emerg Med. 2009;16(1):51–5.
- 5. Pirkei Avot 2:5 (Book of Principles) 200-220 AD.
- Iserson KV. Ethical principles—emergency medicine. Emerg Med Clin North Am. 2006;24(3):513–45.
- Arras JD. A method in search of a purpose: the internal morality of medicine. J Med Phil. 2001;26:643–62.
- Kuczewski M. Casuistry and principlism: the convergence of method in biomedical ethics. Theor Med Bioeth. 1998;19:509–24.
- Beauchamp TL, Childress JF. Principles of biomedical ethics. 1st ed. New York, NY: Oxford University Press; 1979.
- Iserson KV, Biros MH, James HC. Challenges in international medicine: ethical dilemmas, unanticipated consequences, and accepting limitations. Acad Emerg Med. 2012;19(6):683–92.
- Schloendorff v Society of New York Hospital, 105 NE 92, 93, 1914.
- 12. Drane JF. Competency to give an informed consent. JAMA. 1984;252(7):925–7.

- Iserson KV. The three faces of "yes": consent for emergency department procedures. Am J Bioeth. 2007;7(12):42–5.
- 14. Iserson KV. A simplified prehospital advance directive law: Arizona's approach. Ann Emerg Med. 1993;22(11):1703–10.
- Iserson KV. Nonstandard advance directives: a pseudoethical dilemma. J Trauma. 1998;44(1):139–42.
- Iserson KV. Non-standard advance directives in emergency medicine: what should we do? J Emerg Med. 2018;55(1):141–2.
- Iserson KV. More on an unconscious patient with DNR tattoo. N Engl J Med. 2018;378(9):876.
- Iserson KV. The rapid ethical decision-making model: critical medical interventions in resource-poor environments. Camb Q Healthc Ethics. 2011;20(1):108–14.
- Iserson KV, Sanders AB, Mathieu DR, editors. Ethics in emergency medicine. 2nd ed. Tucson, AZ: Galen Press, Ltd; 1995.
- Iserson KV, Goffin F, Markham JJ. The future functions of ethics committees. HEC Forum. 1989;1(2):63–76.
- Iserson KV. Withholding and withdrawing medical treatment: an emergency medicine perspective. Ann Emerg Med. 1996;28(1):51–5.
- Iserson KV. Dangers of withholding treatment in emergency and prehospital settings. Am J Bioeth. 2019;19(3):47–8.
- Boisaubin EV, Lynch GR, Dresser R. Hypercalcemia of advanced malignancy: decision making and the quality of death. Am J Med Sci. 1991;301(5):314–8.
- 24. Popularized by Dr. Edward Livingston Trudeau (1848–1915), but also cited as a 15th century folk saying.
- Spivey WH, Abramson NS, Iserson KV, MacKay CR, Cohen MP. Informed consent for biomedical research in acute care medicine. Ann Emerg Med. 1991;20(11):1251–65.
- 26. Iserson KV. Has emergency medicine research benefited patients? An ethical question. Sci Eng Ethics. 2007;13(3):289–95.
- Iserson KV. Physician ethics in human research: the role of medical publications. Ann Emerg Med. 1990;19(7):828–30.
- Bounes V, Dehours E, Houze-Cerfon V, Vallé B, Lipton R, Ducassé JL. Quality of publications in emergency medicine. Am J Emerg Med. 2013;31(2):297–301.
- Landesman BM. Physician attitudes toward patients. In: Iserson KV, Sanders AB, Mathieu DR, et al., editors. Ethics in emergency medicine. 2nd ed. Tucson, AZ: Galen Press, Ltd; 1995. p. 350–7.
- Moskop JC, Iserson KV. Triage in medicine—part II: underlying values and principles. Ann Emerg Med. 2007;49(3):282–7.
- 31. Kubler-Ross E. On death and dying. New York, NY: Macmillan; 1969.
- Novack DH, Detering BJ, Arnold R, Forrow L, Ladinsky M, Pezzullo JC. Physicians' attitudes toward using deception to resolve difficult ethical problems. JAMA. 1989;261(20):2980–5.
- Iserson KV. Grave words: notifying survivors about sudden unexpected deaths. Tucson, AZ: Galen Press, Ltd.; 1999.
- Iserson KV. Pocket protocols: notifying survivors about sudden unexpected deaths. Tucson, AZ: Galen Press, Ltd.; 1999, 2001.
- Korsch BM, Negrete VF. Doctor-patient communication. Sci Am. 1972;227(2):66–74.
- Iserson KV. Principles of biomedical ethics. Emerg Med Clin North Am. 1999;17(2):283–306, ix. https://doi.org/10.1016/ s0733-8627(05)70060-2.
- Miles SH. The hippocratic oath and the ethics of medicine. New York: Oxford University Press; 2004. p. xiii–xiv.
- Buchanan AE. The question of competence. In: Iserson KV, Sanders AB, Mathieu DR, editors. Ethics in emergency medicine. 2nd ed. Tucson, AZ: Galen Press, Ltd.; 1995. p. 51–6.
- Iserson KV. An approach to ethical problems in emergency medicine. In: Iserson KV, Sanders AB, Mathieu D, editors. Ethics in emergency medicine. 2nd ed. Tucson, AZ: Galen Press, Ltd.; 1995.

Health Disparities

Marcia A. Perry and Joel Moll



65

Introduction

The impacts of inequities and social determinants of health that lead to healthcare disparities and leave many patients vulnerable to poor outcomes have been well documented in patient care. Factors such as access to care, societal bias and stigmatization, lack of autonomy, and social determinants of health, among others, contribute negatively to the effectiveness of the care provided.

The National Academy of Medicine 2002 report *Unequal Treatment: Confronting Racial and Ethnic Disparities in Health Care* concluded that minority patients generally receive lower quality healthcare than white patients in the United States, regardless of insurance status or ability to pay [1]. The report noted that these disparities in quality of care were associated with more deaths among racial minorities than whites. These findings are part of the impetus to examine how structures, processes, and behaviors in our healthcare system may be contributing to disparities among multiple vulnerable groups.

Concepts and Definitions

Healthy People 2020 defines Health Equity as: "The attainment of the highest level of health for all people. Achieving this requires valuing everyone equally with focused and ongoing societal efforts to address avoidable inequities, historical and contemporary injustice, and the elimination of health and healthcare disparities" [2]. Health inequities are differences in health that are not only unnecessary and avoidable but are considered unfair and unjust. Health inequities are rooted in social

J. Moll (🖂)

injustices that make some population groups more vulnerable to poor health than other groups. As a result of health inequities, disparities arise in the presence of disease, health outcomes, or access to healthcare and therapies between population groups. Healthcare disparities are related to provider-patient relationships, provider and systemic bias, social determinants of health, and patient variables such as mistrust of the healthcare system due to past experience. The Minority Stress Model explains that minority groups experience chronic high levels of stress from stigmatization that can contribute to poor outcomes and disparities [3, 4].

Provider and systemic bias can either be explicit or implicit. Explicit bias is often easier to recognize as the outward expression of prejudice. Implicit bias, the unconscious attribution of qualities or values to a member of a certain group, can be much more difficult to recognize. These biases are shaped by experiences and based on learned associations between particular qualities and social categories. Many physicians have different exposures and backgrounds than the patients they serve. Unconscious attitudes and stereotypes affect behaviors and decisions in healthcare. In order to have greater awareness of bias in patient care, it is critical for physicians to receive education, feedback, and coaching on equitable practice.

The population of the United States is becoming increasingly more diverse, with racial and ethnic minority populations predicted to surpass the white majority by 2050. It is critically important that healthcare professionals are educated specifically to address issues of culture in an effective manner that will lead to equitable patient care. Emergency department (ED) patients are from multiple different identities and cultures. Achieving cultural competence in each culture and sub-culture is challenging and not feasible. Cultural *humility* is a more feasible and achievable goal for clinicians [5, 6]. Cultural humility is a lifelong process of self-reflection and self-critique that can inform understanding of cultural differences and how differences require sensitive approaches to healthcare. It requires that clinicians acknowledge how history, culture, and community intersect in the patient's

M. A. Perry

Department of Emergency Medicine, University of Michigan, Ann Arbor, MI, USA

Department of Emergency Medicine, Virginia Commonwealth University, Richmond, VA, USA e-mail: joel.moll@vcuhealth.org

K. H. Todd et al. (eds.), Oncologic Emergency Medicine, https://doi.org/10.1007/978-3-030-67123-5_65

experience. By understanding the power of storytelling and the power of intent, and by active listening and asking deeper questions, we gain a true understanding of the depth of the determinants of health so that we can provide the equitable care patients need and deserve.

Disparities and Cancer Care for Vulnerable Populations

Many vulnerable populations access the healthcare system through the ED, providing both an opportunity and obligation to providers in this setting to deliver equitable care. Patients who present to the ED and are diagnosed with cancer tend to have more advanced conditions, leading to worse outcomes [7, 8]. In one study, patients initially diagnosed with cancer in the ED were 75% more likely to present at stage 4, and 176% more likely to die within 2 years of presentation than those receiving a cancer diagnosis in other settings [7]. ED-diagnosed patients were more often black or from impoverished urban areas [7].

The American Society of Clinical Oncologic has published policy statements outlining the following strategies to eliminate cancer health disparities [9].

- Access to quality healthcare address economic barriers for minorities and the underinsured as well as healthcare delivery.
- Improve healthcare provider, public, and policy maker awareness of racial and ethnic disparities in access to cancer care and outcomes.
- Diversify the oncologic workforce.
- Conduct research to understand the differences in quality of cancer care provided to underserved and ethnically diverse population compared to whites.
- Diversify clinical trials.
- Enhance patient involvement in their care.

Application of these strategies to emergency medicine can help reduce these inequities. As advocates for our patients, we should work consciously to decrease bias, be intentional in diversifying our work force, engage patients in clinical trials, conduct healthcare disparity research, and use cultural humility to provide patient-centered equitable care.

The following cases illustrate concepts of healthcare disparities in vulnerable populations.

Case Study 1

A 56-year-old transgender female arrives to the ED with non-traumatic mid lower back pain. There is no associated weakness, parethesias, or bowel or bladder dysfunction. There is no fever, no past history of intravenous drug use (IVDU). Patient denies medications except estrogen and spironolactone as gender-affirming therapy. Patient was seen and discharged with ibuprofen.

She returns 2 weeks later as the back pain has worsened, and she has now noted some blood in her urine. A clean catch UA shows >100 RBCs and 5 WBCs. She denies dysuria but has some hesitancy. She is placed on nitrofurantoin and discharged.

She returns a week later very uncomfortable and unable to urinate with bilateral flank pain. Bladder scan reveals a markedly distended bladder. She is much improved after decompression with a catheter. A renal CT scan shows multiple lytic lesions in the lumbar spine and a markedly enlarged and irregular prostate, findings consistent with metastatic prostate cancer.

Sexual minorities are a diverse group of individuals without a strict heterosexual identity (lesbian, gay, bisexual, queer or questioning) and/or whose gender identity is different or fluid from their sex-based gender. It is important to recognize that someone's sexual or gender identity may be different from their expression or their behavior. One estimate is that 4.5% of Americans identify as a sexual minority [10] and at least 0.6% as transgender [11]. However, 8.7% of women and 8.2% of men have had same-sex sexual behavior [12].

Sexual minorities have specific health needs intrinsic to their sexual orientation, gender identity, and behavior, in addition to healthcare requirements of the general population. Sexual minority patients face several barriers that may contribute to healthcare disparities, affect their visibility, and impact their relationship with physicians. Historical mistreatment and discrimination, combined with current implicit and explicit bias, can adversely impact the ED experience or decision to seek care at all. Several studies have reported that sexual minorities commonly have negative experiences in the ED, especially in the transgender community [13, 14]. Healthcare benefits and rights may not be extended to sexual minorities. There is not protection against job discrimination in many states, nor at the federal level. Many sexual minorities fear that disclosure of their identity may affect their employment and also their healthcare coverage. Sexual minorities consistently have less access to healthcare, are less likely to be insured, and are more likely to live in poverty.

Transgender patients may have additional identity barriers. They may be listed as their gender assigned at birth in the health record, rather than their gender identity. Many transgender patients do not have identity documents with their preferred identity, with the biggest barrier being cost [15]. This can lead to awkward interactions when the patient appears different than the gender listed in electronic health records (EHR). Importantly for the patient in the case above, the EHR and provider may not have triggered reminders for prostate screening if the patient was listed as female. The patient may not have had gender-affirming surgery to remove her penis, and even if so, most surgeries do not remove the prostate.

Most emergency physicians receive little if any training on sexual minority health in residency [16] and have limited knowledge from medical school [17]. Emergency medicine residents often do not ask about sexual behavior or identity when pertinent, such as the evaluation of an abdominal or GU complaint [18]. Practicing emergency medicine physicians demonstrate poor knowledge of transgender health [19]. Although limited, research suggests that knowledge gaps contribute to important unmet needs in sexual minority patients with cancer [20, 21].

Competent care of sexual minority patient involves not only knowledge, but importantly communication and attitude. When approaching a sexual minority patient with a known or possible diagnosis of cancer, communication and inclusion are important initial considerations. Using language and terminology that is open and not inadvertently offensive is paramount to establishing a trusting and effective relationship. An inclusive environment should be established by using neutral terminology rather than assuming heterosexual and gender concordant identity. Patients should be permitted to include their spouse or significant other if desired, as with any heterosexual couple. When in doubt, ask the patient to define relationships and preferred name.

When taking a history, it is important to focus on information needed for competent care of the patient and avoid irrelevant questions. In the above case, it is clearly relevant to inquire whether the patient has had gender-affirming surgery as it may relate to their presentation or management of genitourinary complaints. In addition, placement of a catheter may be more complex if the penis has been removed and the area has been reconstructed. But for the same patient with an ankle sprain, asking about gender-confirming surgery is more likely curiosity than medical necessity. Similarly, performing a physician exam that is unnecessary due to inappropriate focus is not only unethical but a violation of the patient's trust and would be a damaging experience.

Sexual minorities have been largely overlooked in research on cancer disparities. Disparities have been found with cancers associated with viral infections in men who have sex with men (MSM) [22], including elevated risk for anal cancer, especially in HIV-positive MSM [23]. Although it is postulated that lesbians may have higher rates of breast and reproductive cancer due to possible increased nulliparity, lack of routine collection of sexual orientation data has made determination problematic [24]. Gender-affirming hormone therapy has not been definitely linked to increased breast and reproductive cancer risk in a small number of studies [25]; however, there is some suggestion that when it occurs, breast cancer in transgender females may occur at a younger age (median 51.5 years) [26]. Sexual minorities have higher rates of tobacco abuse, and limited data shows this may translate into higher rates of lung cancer [27, 28].

Evaluation and management of a sexual minority with known or possible cancer diagnosis in the ED can add additional complexities and requires basic competency in treatment of this patient population. Competency in sexual minority healthcare requires the acquisition of knowledge and skills, as well a positive therapeutic attitude, to provide equitable and optimal care.

Key Points

- Sexual minority patients are a diverse group along a continuum of sexual orientation not strictly heterosexual and/ or gender identity different than gender assigned at birth.
- Sexual minorities face many healthcare and legal barriers than can adversely affect access to care and may adversely affect outcomes.
- Emergency physicians receive little training on healthcare of sexual minorities. Basic understanding of healthcare needs, barriers, terminology, and inclusive communication are important to deliver more equitable care.

Case Study 2

A 48-year-old black male with no significant past medical history presented to an urgent care center for rectal bleeding. He is self-employed and has no health insurance. He plows snow in the winters and does lawn care in the summers.

He had stable vitals, has a normal hemoglobin, and was diagnosed as having hemorrhoids after a small external hemorrhoid tag was noted on exam. He was prescribed a cream to shrink the hemorrhoid and was discharged. He has intermittent small amounts of blood in his stool over the next 2 years.

Two years later he presented his local ED stating that his "hemorrhoid was acting up," and he was prescribed a rectal suppository for internal hemorrhoids after a small hemorrhoid tag was again noted. He was told to follow up in the gastrointestinal (GI) clinic if bleeding returns. He called the GI clinic a month later to schedule an appointment and was told they would not see him without insurance and told him to go to the county hospital clinic. The county hospital clinic had no appointment for 6 months. During that 6-month period, he lost his mother, and his adult daughter moved back home, so he missed his appointment due to multiple competing home issues.

One year later, a family member visiting from out of town noted that he had lost weight and was jaundiced, and convinced him to go to the urgent care center. He was sent from the urgent care center to the ED for work-up of anemia and jaundice. After 6 hours in the ED, he was diagnosed with metastatic cancer, which was later confirmed as metastatic colon cancer. He underwent two rounds of chemotherapy before his death.

Colorectal cancer is the third leading cause of cancer deaths in the United States for both men and women. It is largely preventable by timely screening of at-risk groups. The gold standard screening tool, screening colonoscopy, was previously recommended by the American College of Physicians for individuals 50 and over. Due to a recent increase in young individuals under age 50 with colorectal cancer, and a significantly higher risk of colorectal cancer in blacks, it was further recommended that blacks be screened starting at age 45 years. Recently, the American Cancer Society changed screening recommendations to start at age 45 in all individuals irrespective of race and ethnicity. The patient above was not initially referred for colorectal cancer screening and subsequently encountered significant barriers to diagnosis. Minority groups have a lower rate of endoscopic screening recommendations [29]. Socioeconomic status and access to healthcare help account for racial and ethnic disparities in colorectal cancer screening [29]. Patients are more likely to be referred for endoscopy if over age 55, white, with a family history of colon cancer, married, more education, an income greater than \$65K, more frequent doctor visits, and female. Hispanic patients were 34%, and blacks 26% less likely than whites to receive a referral for screening colonoscopy. In summary our patient finds himself in the at-risk group for disparate referral for several reasons: he is under 50, is black, has a low education level, is in the low-income category, has no insurance, and is male.

Along with the new recommendations for a lower threshold for screening to age 45, improved education of emergency physicians on risk factors for colorectal cancer in vulnerable populations as described above is needed to lessen cancer healthcare disparities. This patient had several missed opportunities to receive referrals for screening and avoid what may have been a preventable death.

Key Points

- Colorectal cancer screening should begin at age 45.
- Referral for colorectal cancer screening should be done from the ED for at-risk patients.
- At-risk patients can be identified through social determinants of health questions.

Case Study 3

A 26-year-old female with a history of schizophrenia presents to the ED. She was previously seen in the ED at age 18 after sexual assault and was examined and treated. Eight months later she presented with a complaint of pregnancy and not feeling the baby moving. She had a flat abdomen and a negative pregnancy test and was discharged with a diagnosis of paranoid delusions and referred to psychiatric resources. Over the next few years, she was seen and admitted several more times for delusional psychosis. She was then managed well in the outpatient psychiatric clinic; however, she began presenting again to the ED with abdominal pain and swelling at the age of 24. She had multiple negative pregnancy tests and referrals to psychiatry. Finally she presented to the ED with a court-appointed guardian who insisted on a more thorough medical evaluation because the guardian was convinced that the patient was indeed pregnant. An ED physician who had seen her in the past felt that she had a protuberant abdomen and performed an abdominal ultrasound. Ultrasound revealed a large ovarian mass and the patient was subsequently diagnosed with ovarian cancer.

Patients with mental illness are at increased risk for cancer care disparities. Cancer incidence is similar in individuals with or without schizophrenia. Patients with schizophrenia are 1.5–2 times more likely to die of their cancer than patients without mental illness [30]. Patients with schizophrenia are also more likely to present at advanced stages of disease, are offered surgery at lower rates, and receive fewer chemotherapy sessions [30]. Patients with schizophrenia and other disabilities are often unable to advocate for themselves and are also at increased risk of experiencing physician bias. The 2016 Medscape Lifestyle Report surveyed 15,800 physicians across 25 specialties and found that 40% of physician admitted to bias toward specific types of patients. A higher proportion of emergency physicians reported bias (62%), and that their bias influenced patient care (14%) than any other specialty [31].

Educating emergency physicians regarding unconscious bias with ongoing discussion of methods to mitigate bias is one way to address and decrease the risk of disparate treatment in vulnerable populations.

Key Points

- Patient with schizophrenia are more likely to die of their cancer than patients without mental illness.
- Physician bias exists and emergency physicians admittedly have a high level of bias against patients with mental illness.
- Raising awareness of physician bias and how bias affects clinical decision making is key to helping physicians mitigate bias

Case Study 4

A 24-year-old black female first-year medical student presented to the ED with left chest and breast pain during her first month of medical school. No breast exam was done. Her ED evaluation included a normal chest radiograph and ECG, and she was told to follow up with student health services. At her follow-up she was told it was likely anxiety as experienced often by early medical students. While at home for a holiday break, she saw her primary care physician for persistent left breast pain and a lump was palpated on exam. A biopsy was arranged and she received the unfortunate diagnosis of triple-negative breast cancer. She subsequently underwent left breast mastectomy followed by chemotherapy and radiation therapy.

Breast cancer is more prevalent among white women than black women. However, black women have a higher risk of breast cancer before the age of 40, higher risk of more aggressive cancer, and higher risk of poor outcomes. This largely unexplained disparity in breast cancer outcomes is an area of ongoing research [32]. Williams et al. explored the contribution of social factors (including minority stress) to breast cancer disparities in black women. They conceptualized that increased exposure to psychological stress, low socioeconomic status, and early life adverse exposures that black women experience may be the cause of current disparities in breast cancer incidence and mortality. Increased exposure to psychological stress resulting in epigenetic changes may increase breast cancer risk over one's lifetime. Discrimination and racism as a distinct social exposure experience by racial minorities is an area requiring additional research to separate the factors driven by racism and bias from those driven by socioeconomic status [32].

Key Points

- Black women have a higher incidence of breast cancer than whites before age 40, as well as a higher mortality risk.
- Discrimination, bias, and racism (and other minority stressors) are likely significant contributors to disparity in breast cancer and should be further studied.

Summary

Emergency physicians are trained to work in a high stress, fast-paced environment where the focus is on not missing life-threatening conditions. It is challenging yet necessary for us to find time to address social determinants of health and other barriers to good health and good outcomes. As the safety net for vulnerable population, emergency physicians must develop the skills to identity those at risk for healthcare disparities, consider and address their needs, and strive to eliminate healthcare disparities. Education, research, and advocacy are important priorities for health equity.

References

 Unequal treatment: confronting racial and ethnic disparities in health care. In: Smedley BD SA, Nelson AR, editor. Institute of medicine (US) committee on understanding and eliminating racial

- 2. Office of Disease Prevention and Health Promotion. Healthy people 2030. http://www.healthypeople.gov
- Dohrenwend BP. Social status and psychological disorder: an issue of substance and an issue of method. Am Sociol Rev. 1966;31(1):14–34.
- Meyer IH. Prejudice, social stress, and mental health in lesbian, gay, and bisexual populations: conceptual issues and research evidence. Psychol Bull. 2003;129(5):674–97.
- Tervalon M, Murray-Garcia J. Cultural humility versus cultural competence: a critical distinction in defining physician training outcomes in multicultural education. J Health Care Poor Underserved. 1998;9(2):117–25.
- Foronda C, Baptiste DL, Reinholdt MM, Ousman K. Cultural humility: a concept analysis. J Transcult Nurs. 2016;27(3):210–7.
- Livingood WC, Smotherman C, Lukens-Bull K, Aldridge P, Kraemer DF, Wood DL, et al. An elephant in the emergency department: symptom of disparities in cancer care. Popul Health Manag. 2016;19(2):95–101.
- Rogers MJ, Matheson LM, Garrard B, Mukaro V, Riches S, Sheridan M, et al. Cancer diagnosed in the emergency department of a regional health service. Aust J Rural Health. 2016;24(6):409–14.
- Goss E, Lopez AM, Brown CL, Wollins DS, Brawley OW, Raghavan D. American society of clinical oncology policy statement: disparities in cancer care. J Clin Oncol. 2009;27(17):2881–5.
- Gallup. News. Politics. Newport F. In U.S., Estimate of LGBT population rises to 4.5%. 22 May 2018. https://news.gallup.com/ poll/234863/estimate-lgbt-population-rises.aspx. Accessed 10 Oct 2020.
- Carabez R, Pellegrini M, Mankovitz A, Eliason M, Scott M. Does your organization use gender inclusive forms? Nurses' confusion about trans* terminology. J Clin Nurs. 2015;24(21–22):3306–17.
- Twenge JM, Sherman RA, Wells BE. Changes in American adults' reported same-sex sexual experiences and attitudes, 1973–2014. Arch Sex Behav. 2016;45(7):1713–30.
- Samuels EA, Tape C, Garber N, Bowman S, Choo EK. "Sometimes you feel like the freak show": a qualitative assessment of emergency care experiences among transgender and gender-nonconforming patients. Ann Emerg Med. 2018;71(2):170–82, e1.
- Bauer GR, Scheim AI, Deutsch MB, Massarella C. Reported emergency department avoidance, use, and experiences of transgender persons in Ontario, Canada: results from a respondent-driven sampling survey. Ann Emerg Med. 2014;63(6):713–20. e1
- James SE, Herman JL, Rankin S, Keisling M, Mottet L, Anafi M. The report of the 2015 U.S. transgender survey. Washington, DC: National Center for Transgender Equality; 2016.
- 16. Moll J, Krieger P, Moreno-Walton L, Lee B, Slaven E, James T, et al. The prevalence of lesbian, gay, bisexual, and transgender health education and training in emergency medicine residency programs: what do we know? Acad Emerg Med. 2014;21(5):608–11.
- Obedin-Maliver J, Goldsmith ES, Stewart L, White W, Tran E, Brenman S, et al. Lesbian, gay, bisexual, and transgenderrelated content in undergraduate medical education. JAMA. 2011;306(9):971–7.
- Moll J, Krieger P, Heron SL, Joyce C, Moreno-Walton L. Attitudes, behavior, and comfort of emergency medicine residents in caring for LGBT patients: what do we know? AEM Educ Train. 2019;3(2):129–35.
- Chisolm-Straker M, Willging C, Daul AD, McNamara S, Sante SC, Shattuck DG 2nd, et al. Transgender and gender-nonconforming patients in the emergency department: what physicians know, think, and do. Ann Emerg Med. 2018;71(2):183–8, e1.
- 20. Lisy K, Peters MDJ, Schofield P, Jefford M. Experiences and unmet needs of lesbian, gay, and bisexual people with cancer

care: a systematic review and meta-synthesis. Psychooncology. 2018;27(6):1480-9.

- Harding R, Epiphaniou E, Chidgey-Clark J. Needs, experiences, and preferences of sexual minorities for end-of-life care and palliative care: a systematic review. J Palliat Med. 2012;15(5):602–11.
- Boehmer U, Cooley TP, Clark MA. Cancer and men who have sex with men: a systematic review. Lancet Oncol. 2012;13(12):e545–53.
- van der Zee RP, Richel O, de Vries HJ, Prins JM. The increasing incidence of anal cancer: can it be explained by trends in risk groups? Neth J Med. 2013;71(8):401–11.
- Meads C, Moore D. Breast cancer in lesbians and bisexual women: systematic review of incidence, prevalence and risk studies. BMC Public Health. 2013;13:1127.
- Joint R, Chen ZE, Cameron S. Breast and reproductive cancers in the transgender population: a systematic review. BJOG. 2018;125(12):1505–12.
- Hartley RL, Stone JP, Temple-Oberle C. Breast cancer in transgender patients: a systematic review. Part 1: male to female. Eur J Surg Oncol. 2018;44(10):1455–62.

- 27. McCabe SE, Matthews AK, Lee JGL, Veliz P, Hughes TL, Boyd CJ. Tobacco use and sexual orientation in a national cross-sectional study: age, race/ethnicity, and sexual identity-attraction differences. Am J Prev Med. 2018;54(6):736–45.
- Boehmer U, Miao X, Maxwell NI, Ozonoff A. Sexual minority population density and incidence of lung, colorectal and female breast cancer in California. BMJ Open. 2014;4(3):e004461.
- Ahmed NU, Pelletier V, Winter K, Albatineh AN. Factors explaining racial/ethnic disparities in rates of physician recommendation for colorectal cancer screening. Am J Public Health. 2013;103(7):e91–9.
- Irwin KE, Henderson DC, Knight HP, Pirl WF. Cancer care for individuals with schizophrenia. Cancer. 2014;120(3):323–34.
- Peckham C. Medscape lifestyle report 2016: bias and burnout. 16 Jan 2016. https://www.medscape.com/slideshow/lifestyle-2016overview-6007335. Accessed 10 Oct 2020.
- 32. Williams DR, Mohammed SA, Shields AE. Understanding and effectively addressing breast cancer in African American women: unpacking the social context. Cancer. 2016;122(14):2138–49.
Emergency Oncology in the United Kingdom

Tim Cooksley

Introduction

In the United Kingdom (UK), over 350,000 new cases of cancer are diagnosed a year, contributing to around 28% of UK deaths [1]. In the UK, patients with cancer account for 15% of all acute inpatient stays and its delivery consumes nearly half of the spending on patients with cancer [2].

Cancer care has become increasingly specialised, and advances in therapy have resulted in a larger number of patients receiving care as an outpatient. As a result of these advances in care and the increasing number of patients receiving cancer therapies, there has been a significant increase in the number of patients presenting with unscheduled cancer-related emergencies [3].

The management of emergency oncologic patients presents many challenges. There are toxicities of targeted and immune therapies with which acute care physicians may not be familiar. Early recognition of acutely unwell cancer patients at risk of clinical deterioration is important not only to instigate treatment but also to facilitate decisions regarding whether escalation of care and cardiopulmonary resuscitation is appropriate [4]. This requires an understanding of the patient's underlying prognosis and goals of care, which often requires oncological advice.

In the UK, there have been two strategies adopted to improve the care of acutely unwell cancer patients – the development of specialist admission units in tertiary cancer units and the evolution of "acute oncologic services" to support patients admitted to non-cancer hospitals.

T. Cooksley (🖂)

Acute Oncologic Services

In 2009, following a series of reports recognising that a significant proportion of cancer patients presenting to UK Emergency Departments (ED) received sub-optimal care, the UK National Chemotherapy Advisory Group (NCAG) recommended that every UK hospital with an ED established an acute oncologic service [5]. As a result of the varying demands and resources available across the UK to develop acute oncologic services at each hospital, there has been a wide range of models employed to deliver this strategy. The core of all acute oncologic teams has been an acute oncologic specialist nurse coordinating the service often supported by visiting oncologists.

The fundamental principles of an acute oncologic service are to promote education, awareness and early access to specialist oncologic input. It aims to drive integrated working between acute care physicians, surgeons, medical specialists and oncologists. Acute oncologic supports the variety of ED presentations from initial diagnosis, treatment complications and end-of-life issues. These encompass the three clearly defined types of acute oncologic presentation:

- Type 1 Patients who present with a new diagnosis of cancer
- Type 2 Patients who present with toxicities related to cancer treatments
- Type 3 Patients who present symptoms and complications related to the cancer itself

Despite many national initiatives targeting early diagnosis of cancer in the UK, around a quarter of new cancers continue to be diagnosed during an emergency admission [6]. These patients traditionally were at risk of poorly coordinated care with late referrals to oncologic and palliative care services. Acute oncologic services have played a fundamental role in supporting the management of these patients and ensuring diagnostic pathways are completed in a timely fashion. This is especially pertinent in patients presenting



Department of Acute Medicine, The Christie NHS Foundation Trust, Manchester, Lancashire, UK

with malignancy of unknown origin who often experienced fractured diagnostic journeys with lack of continuity and clinical accountability [7].

A yearly review of a regional network acute oncologic service covering seven hospitals in the North West of England reported 3013 new patient admissions, of which 19% were type 1 admissions, 30% type 2 and 51% type 3 [8]. Acute oncologic models reduced the length of inpatient hospital stays and delivered significant cost savings [8, 9].

Specialist Cancer Admission Units

Case Study

The Christie National Health Service (NHS) Trust is a tertiary oncologic hospital and is one of the largest in Europe. It has 250 beds including a 21-bed Oncologic Assessment Unit (OAU) and a 7-bed Critical Care Unit. It is the lead cancer centre for patients in Greater Manchester and Cheshire but provides many national services.

The OAU is a hybrid urgent care centre/observation unit based on the UK Acute Medical Unit (AMU) model of care, and 30–35% of patients are discharged directly with the rest admitted to downstream inpatient wards [10]. The OAU does not admit patients with symptoms suggesting an acute cardiac event or those who may require emergency surgery. These patients are diverted to their local ED.

A general AMU acts as a 24/7 hub for all emergency medical admissions to hospital and provides a gateway to medical specialties, including oncologic. Its core processes are similar to those in an ED including initial assessment by a competent clinician, early review by a senior clinician, diagnosis with early access to diagnostic tests, assessing and stabilising physiological instability (for a period of up to 48 hours), care delivered by a specialist multi-disciplinary team and triage to appropriate downstream wards if the patient requires an anticipated inpatient stay of greater than 48 hours.

The OAU currently admits around 450–550 patients a month. Patients are admitted to the unit through three main routes:

1. Via a hotline/paramedics – All patients receiving treatment at The Christie have access to a specialist helpline, run by nurse specialists. Patients are advised that if they develop symptoms, such as fever post-chemotherapy, they contact the hotline for advice and assessment. If they are triaged as having a condition related to the cancer or its treatment, they are admitted to the admission unit for assessment. This is often facilitated by the hotline contacting an ambulance to transfer the patient to the hospital.

- 2. Via inpatient clinics/chemotherapy/radiotherapy.
- 3. Via referrals from other hospitals Patients under the care of The Christie or with an acute cancer presentation at another hospital who need urgent chemotherapy/specialist inpatient care are referred to The Christie and transferred to the admission unit.

The OAU is staffed and supported by acute care physicians and advanced nurse clinicians with expertise in emergency oncologic presentations, medical and clinical oncologists, haematologists, supportive and palliative care physicians, visiting medical specialists with interests in complications of cancer therapy, experienced acute oncologic nurses and allied health-care professionals. This model facilitates timely and high-quality tertiary acute oncologic care with a focus on personalised emergency cancer care. It also facilitates innovations that are essential to the delivery of emergency oncologic care, such as triage nurse-led delivery of first-dose intravenous antibiotics in patients presenting with sepsis [11]. One of the strengths of our centre is significant acute care physician and specialist experience in managing acutely unwell patients with immune-mediated toxicity alongside oncologic colleagues, as well as managing the multitude of medical presentations. This ensures high-quality outcomes for acutely unwell cancer patients.

Ambulatory Care

Ambulatory care is recognised as a key tenet in ensuring the safety and sustainability of acute care services. The fundamental basis for ambulatory care is that patients presenting with acute illnesses can be stratified as low risk for developing complications and therefore do not require traditional inpatient care [12]. The NHS targets that 25% of all acute medical presentations are managed through this route.

There are an increasing number of acute cancer presentations that can be risk assessed for care in an emergency ambulatory setting. These include low-risk febrile neutropenia, incidental pulmonary embolism, cancer-associated DVT, chemotherapy-related acute kidney injury, chemotherapy-induced nausea and vomiting, indwelling line infections, acute management of pain crises, malignant hypercalcaemia and other electrolyte abnormalities, asymptomatic brain metastases and malignant pleural effusion [13–15].

Ambulatory models offer the opportunity to integrate palliative and supportive care with oncologic and acute services. Ambulatory enhanced supportive care models have shown utility in the management of low-risk febrile neutropenia [16]. This appears to facilitate improved access for patients to expertise in cancer care and immediate management of the complications of cancer treatment with the goal of preventing downstream complications and future emergency presentations.

Modelling of ambulatory emergency oncologic services with integrated expert supportive and palliative care services is key for providing high-quality, personalised and sustainable emergency oncologic care. It enables a greater number of patients to have their cancer complications managed at their cancer treating centre and aims to reduce attendances at overcrowded general EDs.

UK Role in International Emergency Oncologic Research

The optimal medical management of many cancer-related emergencies is a key area for further research. Many practice patterns are based on expert opinion or prior experience rather than clinical trials. These include traditional presentations such as the management of opioid-related constipation and rescue therapy for chemotherapy-induced nausea and vomiting.

Prospective trials into the emergency management of immune-mediated toxicities of immune checkpoint inhibitors examining optimal doses of steroids for those presenting with life-threatening toxicities and the timing of steroidsparing agents, such as infliximab, are essential. It is key that these studies are not only multi-national but supported by acute care physicians working in emergency oncologic settings.

The care model used for patients with oncologic emergencies needs to be tailored to the local medical and oncologic environment; therefore, it naturally follows that different medical systems have developed different processes to care for these patients. A key for successful emergency oncologic models is the underlying goal of care being provided to these patients by clinicians who are knowledgeable about their needs and have integrated communication with the primary oncologists. Acute care of the oncologic patient is gaining recognition as an important international area that could be improved upon with increased training, research and emphasis on integration into the oncologic system. International collaboration is needed to achieve this.

COVID19 and Acute Cancer in the UK

At the time of writing, the impact of COVID19 on emergency oncologic is unclear. The COVID19 pandemic has resulted in the redeployment of many acute oncologic staff into non-oncologic-based roles to help with its management. The impact and duration of these changes is not yet understood. Furthermore, the outcomes relating to acutely unwell cancer patients with COVID19 and their optimal management is a new challenge for those working in emergency oncologic. Modelling so that patients with SARS-CoV-2 are not colocated with those without will have significant implications for the delivery and modelling of acute cancer services in the UK and internationally. The specialist cancer ER in Asan, Seoul, South Korea, has already been reconfigured to become an infectious diseases/COVID19 ER for the foreseeable future.

It is too early to speculate as to the long-term impact on emergency oncologic services of COVID19, but no current chapter relating to this subject would be complete without a brief caveat suggesting that significant changes in modelling may be necessary.

References

- Cancer Research UK. Cancer statistics for the UK. https://www.cancerresearchuk.org/health-professional/cancer-statistics-for-the-uk.
- Mansour D, Simcock R, Gilbert DC. Acute oncology service: assessing the need and its implications. Clin Oncol (R Coll Radiol). 2011;23(3):168–73.
- Berger J, Cooksley T, Holland M. The burden of cancer on the acute medical unit. Clin Med (Lond). 2013;13(5):457–9.
- Cooksley T, Rice T. Emergency oncology: development, current position and future direction in the US and UK. J Support Care Cancer. 2017;25(1):3–7.
- The National Archives (UK). Department of Health. Chemotherapy services in England: ensuring quality and safety. Last modified 30 Mar 2010. Gateway reference 12208. Archived 4 Jan 2013. Accessed 10 Oct 2020.
- Elliss-Brookes L, McPhail S, Ives A, Greenslade M, Shelton J, Hiom S, Richards M. Routes to diagnosis for cancer – determining the patient journey using multiple routine data sets. Br J Cancer. 2012;107(8):1220–6.
- Richardson A, Wagland R, Foster R, Symons J, Davis C, Boyland L, et al. Uncertainty and anxiety in the cancer of unknown primary patient journey: a multiperspective qualitative study. BMJ Support Palliat Care. 2015;5(4):366–72.
- Neville-Webbe HL, Carser JE, Wong H, Andrews J, Poulter T, Smith R, Marshall E. The impact of a new acute oncology service in acute hospitals: experience from the Clatterbridge Cancer Centre and Merseyside and Cheshire Cancer Network. Clin Med (Lond). 2013;13(6):565–9.
- King J, Ingham-Clark C, Parker C, Jennings R, Leonard P. Towards saving a million bed days: reducing length of stay through an acute oncology model of care for inpatients diagnosed as having cancer. BMJ Qual Saf. 2011;20(8):718–24.
- Lasserson DS, Subbe C, Cooksley T, Holland M. SAMBA18 report – a national audit of acute medical care in the UK. Acute Med. 2019;18(2):76–87.
- Mattison G, Bilney M, Haji-Michael P, Cooksley T. A nurse-led protocol improves the time to first dose intravenous antibiotics in septic patients post chemotherapy. Support Care Cancer. 2016;24(12):5001–5.
- Lasserson DS, Harris C, Elias T, Bowen J, Clare S. What is the evidence base for ambulatory care for acute medical illness? Acute Med. 2018;17(3):148–53.

- Cooksley T, Marshall W, Ahn S, Lasserson DS, Marshall E, Rice TW, Klotz A. Ambulatory emergency oncology: a key tenet of future emergency oncology care. Int J Clin Pract. 2020;74(1):e13436.
- Cooksley T, Holland M, Klastersky J. Ambulatory outpatient management of patients with low risk febrile neutropaenia. Acute Med. 2015;14(4):178–81.
- 15. Hamad M, Connolly V. Ambulatory emergency care improvement by design. Clin Med. 2018;18(1):69–74.
- 16. Cooksley T, Campbell G, Al-Sayed T, Lamola L, Berman R. A novel approach to improving ambulatory outpatient management of low risk febrile neutropenia: an Enhanced Supportive Care (ESC) clinic. Support Care Cancer. 2018;26(9):2937–40.

Cancer Pain Management in Low-Resource Settings

Shiraz Yazdani and Salahadin Abdi

Case Study

A 68-year-old female in a low-resource setting presents to a general practitioner clinic with severe back pain. Upon performing a history and physical examination, the provider finds a large mass on her left breast that is highly suspicious for a malignant process. The patient states that she has had the mass for several years but did not seek care due to fear of being shunned by her community for having this abnormality. Further diagnostic testing reveals an invasive ductal carcinoma which has metastasized to the liver and bony structures including vertebral bodies, causing an acute vertebral compression fracture, which explains the patient's severe back pain. The patient begins treatment for her malignancy with a local oncologist but continues to have severe back pain despite conservative measures. Due to lack of available trained personnel, she is unable to undergo vertebral augmentation to treat her compression fracture. Her team attempts to control her pain with a combination of opioids and adjuvant medications. However, due to regional restrictions on opioid importation, her treatment options are limited to tramadol, codeine, and immediate-release morphine sulfate. Healthcare providers attempt to get approval for transdermal fentanyl in order to improve her pain control but are unable due to the cost to the hospital and clinic system for importing this medication. They also find a provider who can perform vertebral augmentation to improve her pain, but that facility is several hours away and the patient states that she cannot afford to travel there. Furthermore, there is a lack of governmental assistance programs to help patients such as her travel to seek more advanced care.

S. Yazdani

S. Abdi (🖂)

Introduction

Despite advances in the treatment of cancer, a large proportion of oncologic patients report disease-related pain at some point during the evolution of their disease process. The incidence of cancer-related pain increases with more advanced disease. Suffering from this pain can significantly impact a patient's activities of daily living and quality of life [1]. Despite advances in the treatment of various cancer disease processes, as well as the numerous etiologies of cancer pain, uncontrolled pain remains a significant issue. The prevalence of pain has been reported as 64% in those patients with advanced disease. Even in cancer survivors, the prevalence has been as high as 33% [2]. Treatment of cancer and cancerrelated pain varies based on geographic differences in availability of information, training of healthcare workers, and availability of resources (equipment for interventional procedures, medications for pharmacologic treatment, appropriately trained providers for non-pharmacologic management, etc.). In certain settings, resources can be limited. This can lead to significant alterations in the availability and applicability of treatments for cancer pain. The vast differences in resources can be attributed to unequal distribution of power, income, goods, and/or services [3]. These differences occur within and between countries. Certain populations have been reported to be at higher risk for such inequalities including those with low-income, minorities, and underserved women [4]. On a global scale, many low- and middle-income countries experience this low-resource environment, and patients in those countries may suffer from untreated or undertreated cancer pain. The primary countries associated with the most readily available resources include Western Europe, North America, and Oceania [5]. Roughly, countries can be classified according to their human development index as compared to other countries. This composite includes components of life expectancy, education, and per capita income [6]. Although diagnosis and treatment in these settings may be challenging, it is critical to strive to provide the same level of care across the board. In an effort to highlight this goal, the



Interventional Pain Management, Lubbock Spine Institute, Lubbock, TX, USA

Department of Pain Medicine, The University of Texas MD Anderson Cancer Center, Houston, TX, USA e-mail: sabdi@mdanderson.org

K. H. Todd et al. (eds.), Oncologic Emergency Medicine, https://doi.org/10.1007/978-3-030-67123-5_67

Diagnosis/etiology

United Nations has released Sustainable Development Goals for 2030. These goals call for a reduction in premature mortality by a third using appropriate prevention and treatment modalities [7]. In this chapter, we highlight cancer pain and its treatment, ranging from etiologies and modalities of treatment to challenges and opportunities in low-resource settings.

Etiologies of Cancer Pain

Understanding and treating the etiologies of cancer pain requires a detailed understanding of the assessment of the cancer pain patient [8]. This is particularly important in the low-resource setting, as patients in these settings can present unique challenges in terms of education levels, expectations, social situations, family dynamics, and cultural belief systems. Beginning at the most basic level of assessment lies the chronicity of pain. Even at this level, patients in lowresource settings may present in a later stage of chronicity due to low access to care [9]. This lower level of access leads to many patients in low-resource settings to be diagnosed at later stages of disease [10], which is associated with higher levels of pain. Additionally, cancer pain is a unique type of pain that often evolves at a rapid pace. It requires consistent re-evaluation and re-assessment. These constant changes may be from a change in the disease process (e.g., tumor growth or metastasis), response to treatment (e.g., reduction of tumor burden), treatment of pain generators (e.g., nerve blocks), adverse reactions to treatment (e.g., chemotherapy-induced peripheral neuropathy), side effects from medications (e.g., myalgias), or a combination of any of these issues. This constant re-assessment requires significant resources but is critical in treatment as the etiology of pain can shift quickly in a patient with cancer (Fig. 67.1).

The most basic place to begin with the etiologies of cancer pain is determining the pain's cause and pathophysiology. Syndromes are often used to describe commonly associated states in cancer pain [11], but a universal classification system for these syndromes has not yet been determined [12]. A simple way to classify different types of pain is whether they are related to tissue damage (somatic nociceptive, visceral nociceptive) or nervous system disorders

Assessment



Fig. 67.1 Barriers to the treatment of cancer pain in low-resource settings

(neuropathic). The majority of cancer pain is directly attributable to tumor burden, with other causes being treatmentrelated or non-cancer-related [13].

Somatic Nociceptive Pain

Pain related to tissue damage is often described as achy or throbbing. This type of pain is often well localized. It can come from a variety of tissues including skin, muscle, and other types of soft tissue. This is the most common type of pain in cancer patients. Oftentimes, it is due to bony disease or metastases [14]. Somatic nociceptive pain generators include various tumor-related sites. This can be from bony disease including primary bone tumors, bone marrow expansion in hematologic malignancies, and bony metastases (e.g., vertebral body metastasis from prostate cancer). Spinal pain can manifest from tumor burden in the vertebral body, bony destruction, vertebral body fractures, and/or spinal cord compression. Soft tissue tumors lead to various pain types at the site of the tumor, including facial pain, headaches, and pleural pain. Further pain etiologies include paraneoplastic pain syndromes such as osteomalacia, osteoarthropathy, and myalgias.

Visceral Nociceptive Pain

In contrast to somatic nociceptive pain, visceral nociceptive pain is often poorly localized. Depending on which organ is affected, it can also be referred to different regions in the body. The characteristics of visceral pain can include a distended feeling and a generalized, dull pain. For example, pain in the myocardium is often referred to the upper extremity. Pain in the gallbladder may be referred to the shoulder. Pain in the ureter may be referred to the lower quadrant. Any organ can be affected but more common sources of visceral pain include hepatic metastases, intestinal obstruction, or venous distension due to tumor burden [15]. Visceral nociceptive pain generators include various syndromes and can result from hepatic distension, intestinal distension, and/or obstruction, peritoneal carcinomatosis, perineal pain, adrenal pain, and ureteral obstruction.

Neuropathic Pain

This third type of pain arises from cancer or cancer treatment affecting the nervous system. This can result in nerve irritation or damage. Abnormal nerve firing may result in dysesthesias, allodynia, hyperesthesias, hypoesthesias, and anesthesia dolorosa. This type of pain is often described as burning, shocking, or lancinating. Neuropathic pain can result from a direct nerve insult (e.g., a tumor compressing a nerve) or inflammatory mediators from adjacent soft tissue destruction. Certain cancer treatments may lead to neuropathic pain states as well, such as chemotherapy-induced peripheral neuropathy or radiation neuritis. Furthermore, surgical treatment of tumors can lead to pain states such as phantom limb syndrome [16]. Certain treatments may also lead to other pain-causing disease states to manifest themselves, such as immunosuppression from chemotherapy leading to post-herpetic neuralgia and subsequent neuropathic pain. Other painful neuropathic states can include leptomeningeal metastases, glossopharyngeal neuralgia, trigeminal neuralgia, various plexopathies (cervical, brachial, lumbar, etc.), peripheral mononeuropathy, malignant radiculopathy, and paraneoplastic sensory neuropathy [17].

Challenges and Opportunities for Cancer Pain Management

The low-resource setting provides numerous challenges and opportunities in the setting of treating cancer pain. The best approach may be to analyze the usual goals and approach for the treatment of cancer pain and then consider the challenges that are presented in a low-resource setting. At that point, the opportunities to optimize care can be elucidated.

The first goal in treatment should be accurate assessment of pain. This can be quite challenging at baseline due to, as previously mentioned, the complex interaction of disease burden, treatment effects, and pain management interventions [18]. Also at play is the fact that non-cancer-related pain etiologies have been reportedly involved in as high as 9% of pain diagnoses [19]. The practitioner should have the patient characterize the pain. This can include temporal features, intensity, quality, relieving and precipitating factors, location, and radiation. Even this simple assessment presents a challenge in the low-resource setting. Due to lack of education, descriptors of pain may be limited. In some settings, lower literacy rates can severely affect accurate assessment. Using functional scales such as the complex disability indices can prove difficult. Lack of trust between certain populations and healthcare workers can lead to inaccurate reporting of pain. This has been reported with indigenous people [20]. An additional confounding factor in determining the etiology of pain in the setting of cancer in low-resource settings is the lack of education and subsequent differences in expectations between the patient, family, and healthcare providers. Communication is essential in order to optimize treatment [21]. The opportunities presented here are self-evident. Increased education to the public can improve communication between patients and healthcare workers. This can also help build trust between the two. Successful public marketing campaigns aimed at improving the image of the healthcare system in general should be a goal. If available, utilizing local trusted figures or champions can help bridge the gap between patients and healthcare workers. Healthcare workers should be well educated in regional differences in social norms, cultural beliefs, and patient-family expectations so that there is not a disconnect when planning the goals of care.

Next, the practitioner should understand the nature of the pain. This not only includes the cause but also the relationship to the pathophysiology of the related cancer or cancer treatment and any associated syndromic features. This presents a challenge as many of these patients may have not undergone appropriate testing due to lack of equipment or financial issues. For example, if a tumor-related vertebral compression fracture is suspected but the area in which the patient and healthcare worker reside does not have access to advanced imaging such as magnetic resonance or computed tomography, the diagnosis cannot be confirmed and treatment may be indefinitely delayed. This is further complicated by the fact that many patients in low-resource settings present at later stages of disease. It has been shown that lower- and middle-income countries have higher mortality rates, leading to significant cancer survival gaps. This is oftentimes due to diagnosis at a later stage and lack of access to prompt treatment (Fig. 67.2) [20, 22].

The challenges here are more systemic in nature and, therefore, more difficult to address. Access to early diagnosis necessitates earlier access to care. This could mean higher numbers of available clinicians to make the diagnosis as well as more healthcare facilities in which to assess, diagnose, and treat a larger number of patients. This also entails facilities with more advanced imaging modalities with which to diagnose patients. Local healthcare workers would also require more nuanced education on the assessment and treatment of cancer and cancer pain. Unfortunately, implementing a large-scale change such as this requires significant funding. Public financing plays a crucial role in this type of implementation. In high-income countries, this type of funding is what has led to modern, sustainable healthcare infrastructures. In most of these countries, universal healthcare is now also standard, leading to increased access to care and health equity across the population. Aid to low- and middleincome countries by countries with higher GDP has been critical in maintaining access to healthcare. However, as income in countries providing aid has increased, the proportion of aid provided, sadly, has not (Fig. 67.3) [3].

Additional challenges in this realm include appropriate access to cancer treatment modalities. Stereotactic radiotherapy, for example, may not be available in a low-resource setting. However, this type of therapy is ideal in some scenarios, such as localized bone tumors [23]. Again, the primary opportunity here comes at great financial cost. Public funding campaigns may be utilized to help fund improvements in cancer treatment modalities. Additional aid from other countries can be sought as well. Some have suggested greater debt relief to low- and middle-income countries so that they can use their government income to improve healthcare infrastructure. Generalized improvements in government infrastructure and strengthened internal tax capacity are systematic improvements that may eventually lead to benefits in the healthcare system.

Once the underlying cause of pain is elucidated, its effects on the patient's quality of life and psychosocial well-being should also be determined and addressed. This can include effects on activities of daily living, physical functioning and capacity, mood, coping, stress levels, familial effects, social interactions, sleep quality, and sexual function. This also ties in with general psychiatric well-being, including anxiety and

Fig. 67.2 Ratio of mortality to incidence in a specific year by cancer type and country income. People with cancer who live in higher-income countries are often less likely to die from their cancer, compared to people who live in lower-income countries. (From Farmer et al. [22], reprinted from the Lancet with permission Elsevier)





Fig. 67.3 The growing gap: per capita aid from donor countries relative to per capita wealth, 1960–2000. Calculated in \$US at 1998 prices and exchange rates. GNI = gross national income. (From Marmot, et al. [3], with permission Elsevier)

depression. Assessment of substance use disorders, personality disorders, and spiritual well-being should also be carried out. This is an area that can prove quite challenging in low-resource settings. Due to lack of education, a cancer diagnosis often carries a stigma and inherent fear factor. There have been reported cases of patients believing it is contagious [24], leading to the patient hiding the diagnosis so that they are not shunned by the community. This, of course, leads to a lack of care until the disease is at a late stage or, in more severe cases, until the patient has passed away from the disease.

Once an accurate diagnosis is made, the treatment plan can begin to be formulated. In an ideal setting, this involves a multidisciplinary team that focuses their efforts on treating the patient, disease process, and symptoms such as pain at the same time. This team can include, but is certainly not limited to, an oncologist to treat the disease of cancer, an internist to treat other underlying medical comorbidities, a psychiatrist to treat coexisting anxiety and/or depression, a physiatrist to coordinate rehabilitative modalities, a pain specialist to perform interventional procedures and optimize pain medication regimens, a radiation oncologist to evaluate for therapeutic radiotherapy, a pharmacist to monitor for medication optimization and interactions, a nutritionist to plan an ideal diet for health and recovery, and even a pallia-

tive medicine specialist or hospice physician in cases of endstage disease. Coordinating such a large team requires considerable resources, which may not be available in limited or low-resource settings. Challenges in these scenarios involve not only availability of specialists, but education in general. Due to limited resources, healthcare providers in these settings may have to play the role of many of these specialists. However, they may lack the education to properly execute these roles. In some cases, the education may be extremely superficial; in others, it may be absent altogether. Opportunities to improve education are vast but require significant long-term commitment from the institutions in that specific infrastructure. There is no short-term solution or "band-aid" for appropriate training. In fact, rushed training may lead to inaccurate diagnoses and improper treatment decisions, leading to patient harm. Appropriate training may require outsourcing, at least on a temporary basis, until educational institutions can provide appropriate teaching and mentorship for trainees.

An additional area which requires education for both the healthcare provider and the patient population is palliative and hospice care. The challenges and opportunities surrounding healthcare provider training are covered above. From the patient population perspective, there are many myths and stigmas present regarding end-of-life care. These may be tied to certain religious or spiritual beliefs. For example, fatalism (the belief that events in life are predetermined) can lead patients and their families to forego treatment, even at the end stages of disease, because they believe what is transpiring is "meant to be." This can lead to suffering from multiple untreated or undertreated symptoms, including pain. At the end of life, all patients deserve to have the best quality of life they can be given. This requires debunking these myths and removing the stigmas surrounding cancer in many low-resource settings. Many campaigns focus on the word "cancer" and exposing the public to it so that it is not a taboo word or diagnosis. Exposing the community to patients who are cancer survivors is also crucial to show the community that cancer is survivable with appropriate diagnosis and treatment. Using local, well-known figures is an effective method to increase exposure and improve education about cancer diagnosis and treatment.

Managing Cancer Pain

Part of the treatment plan centers around appropriate, multimodal pain management. This includes pharmacologic management, including opioids as clinically indicated, nonpharmacologic management, and interventional procedures. Disease-modifying treatments such as chemotherapy and radiotherapy can significantly affect the patient's pain. However, covering the details of these treatments is beyond the scope of the current chapter. Briefly, these treatments can be considered part of the pain management plan if their administration has a viable chance of reducing tumor burden and thereby reducing the effects of pain by the tumor. Of course, these treatments are not without contraindications and adverse effects. Formulating a plan to treat pain with disease-modifying treatments requires careful coordination between the pain specialist, oncologist, and radiation oncologist.

Non-opioid Pharmacologic Management

The World Health Organization released the cancer pain ladder as a schematic for the basis of cancer pain treatment almost 35 years ago. At the first step of the ladder is nonopioid pharmacologic management. This can include nonsteroidal anti-inflammatories and acetaminophen. Non-steroidal anti-inflammatories have been shown to be particularly effective for bony pain from primary or metastatic tumors. This can also be applied to osteolytic lesions. These modalities can also be combined with opioid treatment, although the combination may not be necessarily more effective than opioid monotherapy [25].

These therapies have the benefit of being widely available, even in low-resource settings. However, they come with significant risks. Acetaminophen carries an inherent risk of hepatotoxicity and must be used with caution in patients with compromised liver function. In low-resource settings, one unique consideration is the undertreatment of alcoholism. Alcoholism is a common issue in many regions, and treatment requires long-term rehabilitation - an effort which requires substantial community resources [26]. In settings where resources are limited, alcoholism may be undertreated and treatment with any medications which may compromise hepatic function should be prescribed with caution. Non-steroidal anti-inflammatory drugs have their own set of cautions to consider. They can adversely affect the gastrointestinal, renal, and cardiovascular systems [27]. In lowresource settings, patients should be carefully screened for abnormalities in any of these symptoms prior to treatment with non-steroidal anti-inflammatory drugs. As mentioned earlier in the chapter, patients in low-resource settings avoid seeking healthcare for a multitude of reasons. This presents major issues not only with cancer diagnosis and treatment, but with medical comorbidities as well. If a patient delays seeing a healthcare provider for cancer care, it can be assumed that the same patient likely did the same for other medical problems. Undiagnosed or undertreated hypertension, diabetes, renal disease, coronary artery disease, peripheral vascular disease, and peptic ulcer disease may all be present in patients presenting for care. A thorough history and physical and appropriate diagnostic modalities are critical in ensuring these patients will not be harmed if they are prescribed non-steroidal anti-inflammatory drugs. Some of these risks may be mitigated by using a selective cyclooxygenase-2 inhibitor and/or combining treatment with a proton pump inhibitor [28].

Certain adjuvant medications have been effectively used in conjunction with acetaminophen, non-steroidal antiinflammatory drugs, and opioids. These adjuvants consist of a variety of classes of medications and treat various types of cancer and treatment-related pain. Being non-opioid, their use should be considered in certain low-resource settings in which opioids may be unavailable due to cost, restrictions, or stigmas. Anticonvulsants such as the gabapentinoids (gabapentin and pregabalin) are often used given the high prevalence of neuropathic components in cancer pain. These N-type calcium channel blockers are often well tolerated and widely available. Gabapentin has been in use for decades and is generally available. Pregabalin is newer but is now available in generic form. Other anticonvulsants such as carbamazepine, topiramate, oxcarbazepine, valproate, and lamotrigine may also be used for neuropathic pain. These medications may not be as available as the gabapentinoids and some have the potential for significantly greater adverse effects. Antidepressants, specifically tricyclics and serotonin-norepinephrine reuptake inhibitors, can be effective in neuropathic cancer pain as well. Many of these medications have been widely available for quite some time as well and have low associated costs. They can be particularly helpful in treatment-related neuropathic pain states such as chemotherapy-induced peripheral neuropathy. Another class of medication are topical adjuvants. This can include lidocaine cream, lidocaine patches, lidocaine liquid, capsaicin cream, capsaicin patches, non-steroidal anti-inflammatory cream, doxepin cream, and doxepin elixir. Although many of these are now available in generic form, any medication with a transdermal application system increases the manufacturing cost. Additionally, a patient may require multiple patches. In a low-resource setting, this may not be ideal for pain that may be chronic or expected to continue for the long term. For bone pain, glucocorticoids, bisphosphonates, and calcitonin are often used. Glucocorticoids have a host of adverse effects, especially with long-term use, and should be used with caution. Additionally, their use should be coordinated with the treating oncologist due to risk of immunosuppression. Bisphosphonates are considered generally safe, although there is a risk of osteonecrosis and renal injury. These concerns should be addressed with a careful history and physical examination, the challenges of which are outlined above [29].

Opioid Pharmacologic Management

Opioids remain the mainstay of treatment of cancer pain, particularly in advanced stages of disease. They have significant risks associated, including respiratory depression, nausea, constipation, pruritus, etc. The risk of respiratory depression remains the biggest clinical concern, as this can be life threatening. This risk is compounded by the concurrent use of benzodiazepines. Patients with cancer pain are often prescribed benzodiazepines due to the high incidence of anxiety surrounding a cancer diagnosis as well as cancer treatment. Careful risk stratification is needed to mitigate the risks surrounding opioids. These risks have been highlighted in recent years in the midst of the opioid epidemic. Overuse, misuse, and diversion remain significant concerns in the prescription of opioid analgesics [30]. Patients should undergo a risk assessment before beginning treatment with opioids. This can help identify high-risk patients and guide the clinician in appropriate prescribing for a particular patient. Consistent reassessment should be performed, as patients can shift between risk stratification categories over time. Additionally, patient education is critical when it comes to opioid management. Tempering expectations and explaining the risks of opioid use as well as the differences between addiction, physical dependence, and tolerance are vital in maintaining a healthy provider-patient relationship, espe897

cially in the setting of chronic opioid use. Monitoring patients consistently for appropriate outcomes is also important to ensure that the medication prescribed is helping them achieve their goals and outcomes. Outside of arbitrary pain scores such as the visual analogue scale, functional outcome measures should be utilized. Examples include the Short-Form 36, the Brief Pain Inventory, and the Patient Global Impression of Change [31].

Selection of opioids remains largely dependent on regional availability and clinical familiarity and training. Pure mu-opioid agonists remain the most commonly used and available. Initial selection will also depend on the severity of pain and the patient's tolerance to opioid agonists. In opioid naive patients, a starting point may include tramadol, a weak mu-opioid agonist with serotonin-norepinephrine reuptake inhibition effects or codeine. Tramadol's unique mechanism of action may make it more suitable for patients with a larger component of neuropathic pain. Caution should be used in those patients with renal dysfunction or those who are prone to seizures, as tramadol can reduce the seizure threshold. Codeine's mechanism of action relies on its conversion to morphine. In certain patient populations, hypermetabolism of codeine may lead to potentially toxic doses of morphine [32]. In the next tier of opioids, hydrocodone is most commonly available in North America and, to a lesser extent, in Europe. The immediate-release form is paired with acetaminophen (now limited in the United States to 325 mg acetaminophen per tablet), and the newer extended release form is hydrocodone without any paired medications. Caution should be used, as mentioned previously, with acetaminophen use. The extended release form is newer and currently brand name only, signifying that it is a higher cost and may be more difficult to obtain. In low- and middle-income countries, hydrocodone is scarcely available. Other opioid options include oxycodone, oxymorphone, morphine, and hydromorphone. Each of these is available in short acting, as well as long-acting, formulations. If cancer pain is continuous, a long-acting formulation is preferred with an immediate-release formulation given additionally for breakthrough pain. Caution should be used with hepatic dysfunction in any of these medications, but particularly with morphine due to accumulation of its metabolites [33]. Fentanyl is a potent opioid most commonly available as a transdermal patch for long-acting analgesia. It is also now available in a transmucosal formulation for rapid onset action. However, due to its unique routes of administration, manufacturing costs cause the cost to the patient to oftentimes be higher than other opioids. This is particularly problematic in low-resource settings. Availability of fentanyl products may also be an issue, although the transdermal fentanyl patch has been used long enough now that it is widely available. Methadone is a unique opioid agonist that also has an effect on the N-methyl-D-aspartate receptor. This medication has a long and variable half-life, which can be helpful in

continuous pain but may also pose an issue if it is to be discontinued as it persists in the system for an extended period of time. Its NMDA activity is beneficial when treating pain with a neuropathic component. It also lacks active metabolites, making it relatively safer for those patients with renal dysfunction. It does carry unique considerations, including the potential for QT interval prolongation. This should be monitored closely in order to prevent risk of torsades de pointes. Another issue is the stigma associated with its use. Since it is used in rehabilitative programs for opioid addiction, patients may be hesitant to use it for their cancer pain. In the low-resource setting, stigma with opioid use is already a major issue and the additional stigma with methadone may cause patients to decline treatment with it.

In the cancer pain population, routes of administration of opioids play an important role in treatment. Many types of cancer and cancer treatment may affect gastrointestinal absorption, leading to inadequate analgesia with oral opioids. Consideration should also be given to opioid metabolism and accumulation of metabolites in those patients whose hepatic and renal systems may be affected by disease or treatment. Patients undergoing multiple surgeries may be nil per os frequently, necessitating the use of alternative routes of administration such as intravenous, transdermal, or mucosal. Adjustments in doses may need to be frequent and require timely reassessment. When changing doses, several days may need to elapse for steady state blood concentrations to change with long-acting opioids. This may be up to 1 week in the case of a unique medication such as methadone.

Management of side effects is also crucial in opioid treatment for cancer pain. Opioid induced constipation remains a major issue for many patients, so much so that they may want to decrease doses in order to improve this side effect. Dietary adjustments and over-the-counter stool softeners and stimulant laxatives may be used in mild cases. Peripheral opioid antagonists such as naloxegol may be used for more severe cases. Opioid antagonists such as methylnaltrexone and naloxone may also be considered in select cases. Unfortunately, the availability of these antagonists is highly variable, especially in low- and middle-income countries. Other side effects to consider include mental clouding, somnolence, pruritus, nausea, and urinary retention. These can normally be ameliorated by dose adjustment and adjuvant treatment (e.g., a stimulant for mental clouding and a 5-HT3 antagonist for nausea).

Opioid Consumption in Low- and Middle-Income Countries

Despite significant data indicating that cancer pain continues to be often undertreated, there remain significant barriers in S. Yazdani and S. Abdi

treatment of pain in low-resource settings such as low- and middle-income countries. The barriers to appropriate care are numerous and include low priorities given to pain relief. Additionally, national guidelines on pain treatment are often lacking or outdated. This is accompanied by social stigmas surrounding cancer, cancer treatment, and cancer pain, as highlighted previously in this chapter. On the topic of opioids, availability remains the major issue. This may be due to governmental restrictions on the import of opioids as well as restrictions on use due to fear of misuse, overuse, addiction, and diversion [34]. Policy issues causing decreased availability include lacking national guidelines on opioid imports, tight regulations on import and use due to fear of misuse and cultural beliefs and/or stigmas, and inadequate health governance. Clinical barriers to opioid use include lack of prescribers, lack of education in the healthcare worker community as well as the patient population, stigma and fear, and limited number of clinics available to see the appropriate number of patients and spend enough time with them for opioid education. Systemic resource issues also contribute and may include lack of resources to store and appropriately distribute opioids, low or absent numbers of pharmacies (especially in rural regions), and inconsistent stock.

As morphine equivalents remain the standard by which all opioids are judged, global consumption via morphine equivalents is a metric used by the World Health Organization to study global opioid use (Fig. 67.4) [35].

Low- and middle-income countries accounted for 84% of the world's population in 2018 [36]. The disparity in availability of morphine and other opioids in relation to patients in need is highlighted in the map above, from a report in The Lancet Commissions. This figure is a dismal reminder of the glaring disparity between high-income and low-/middleincome countries. In the United States, Canada, and Western Europe, the population remains over-served by at least hundreds of percentage points greater than their need. In lowand middle-income countries, the population remains underserved usually below 15% and even below 1% in some countries. In addition to the systemic issue related to opioid use in low- and middle-income countries, one additional issue remains inconsistent availability of a medication that a patient may be currently using. Month to month, stock changes may require clinicians to alter the medication used. For example, a patient may be stable on transdermal fentanyl for 3 months, and, the following month, the country no longer has that medication available. Although opioid rotation is an established practice, it is often a slow and inaccurate process and may lead to periods of uncontrolled pain. This is especially troublesome when applied to a patient who was doing well on one medication but now has to switch to one which may be less ideal because the country where they reside does not have policies or procedures in place for stable stocks of medications.



Fig. 67.4 Distributed opioid morphine equivalent (morphine in mg/ patient in need of palliative care, average 2010–2013) and estimated percentage of need that is met for the health conditions most associated

with serious health-related suffering. Source: International Narcotics Control Board and WHO Global Health Estimates, 2015. (From Knaul, et al. [35], with permission Elsevier)

Non-pharmacologic Management

Non-pharmacologic, non-interventional treatments for cancer pain are becoming more commonly used as they offer potential benefits with minimal side effects. This can include psychotherapy, physical therapy, and acupuncture. This category also encompasses programs which may assist with associated components of cancer pain such as pain education, support groups, family stress changes requiring therapy, financial programs to assist with pharmacologic or interventional treatments for pain, etc. Disparities in low-resource settings can significantly affect response to these treatments. One study has shown a favorable response to patient coaching in the management of cancer pain in ethnic minorities [37]. Educational programs are critical in reducing disparities between groups in low-resource settings. As previously mentioned, lack of education leads to propagation of myths and stigmas which delay care and lead to diagnosis in later stages of disease with greater disease-related suffering. Dispelling myths surrounding physical manipulation methods such as physical therapy and acupuncture is also vital, as

these modalities can significantly improve functionality in cancer pain patients. Family counseling is also an important component of cancer pain treatment, especially when approaching the end of life and with involvement of palliative care.

Interventional Procedures

Interventional procedures can play a crucial role in the treatment of cancer pain. As pain interventions have advanced, practitioners have been able to perform more advanced procedures without increasing the invasiveness of these procedures. The types of procedures performed depend on an accurate diagnosis of the underlying cause of pain. Special consideration must be given to any areas surrounding a tumor or metastases, given that these areas are highly vascular and the risk for bleeding may be increased. Additionally, consideration must be given to the patient's oncologic treatments, as certain therapies may interfere with clotting ability and wound healing. The use of corticosteroids must also

For head and neck tumors, peripheral nerve blockade is available as a treatment modality to assist with pain management. Examples of such blocks include trigeminal nerve blocks, stellate ganglion blocks, occipital nerve blocks, and cervical nerve root blocks and/or epidural steroid injections. For upper and lower extremity tumors, somatic nerve blockade can provide temporary relief or pain management surrounding surgical resection. Sympathetically mediated pain can be treated using stellate ganglion or lumbar sympathetic blocks. Post-surgical syndromes such as post-mastectomy pain syndrome and post-thoracotomy pain syndrome can be treated with peripheral nerve blockade, such as intercostal nerve blocks. Splanchnic nerve or celiac plexus blockade (and ablation) is commonly used for visceral abdominal pain. Similarly, superior hypogastric plexus blockade (and ablation) can be used for visceral pelvic pain. Pain at the perineum and rectum can be treated by blocking the distal end of the sympathetic chain, at the ganglion impar.

More advanced therapies include vertebral augmentation, which is used to treat tumor or osteoporosis-related vertebral compression fractures. Newer technologies allow for radiofrequency ablation of tumor sites in the vertebral body prior to polymethylmethacrylate injection. Implantable therapies such as dorsal column stimulation and intrathecal drug delivery systems are available as well. In cancer pain, intrathecal drug delivery systems allow for the use of intrathecal opioids to effectively treat pain while lowering some adverse effects of oral or transdermal opioids. For severe, refractory cases of pain, advanced functional neurosurgical techniques such as cordotomy can be employed.

Most of these interventions are performed under image guidance—commonly fluoroscopy or ultrasound. Computed tomography guidance may be used in select cases.

Barriers to the use of interventional techniques remain present in low-resource settings. The setting and equipment needed to perform such interventions may not be readily available. The cost to set up a procedure suite to perform these interventions is much higher than that of acquiring most pharmacologic treatments. Additionally, the training required for the practitioner remains substantial. Specialists who can perform such procedures may not be readily available in low-resource settings, and systematic infrastructure changes will need to be implemented before this issue can be resolved.

Recommendations

In order to overcome barriers in care in low-resource settings, several changes can be undertaken in a systematic, stepwise fashion in order to improve access to care. The first place to begin is changes in policy. This can occur on a local, regional, and national level. Policies need to be in place to improve access to care for cancer pain patients. This includes dedicating resources, financial and otherwise, to the development of appropriate healthcare facilities, training of healthcare personnel, advocating to local and national community leaders, and working with national and international groups in order to implement these policies. These groups and organizations can be small, local focus groups or large, international organizations such as the Pain & Policy Studies Group [38]. These policies need to be aimed at increasing availability of care for cancer pain patients and avoiding restrictions usually placed on issues such as opioid importing and regulations.

Availability of pharmacologic treatments is another area that is high yield in terms of increasing care to cancer pain patients. In addition to the policy changes mentioned above, infrastructure needs to be put in place to obtain, store, and properly dispense medications to alleviate pain and suffering. This includes having licensed healthcare providers who are well trained in these treatments and are able to dispense them to patients appropriately. Consistency also needs to be maintained on medication stocks so that fluctuations do not occur regularly. In countries with tight regulations on opioids in particular, exceptions should be made for those suffering from cancer pain so their treatment is not delayed or prevented.

Improving education is critical in improving care for these patients. This applies to both healthcare workers and the population in general. In regard to healthcare workers, they need to be familiar with the tenets of cancer pain treatment and feel comfortable applying principles commonly used as the standard of care around the world. There also need to be enough healthcare workers so that a small number do not get overwhelmed by a large patient-to-healthcare worker ratio. A high ratio can lead to inadequate time with the patient, increasing the potential for misdiagnosis and inappropriate treatment choices. An example of healthcare worker education occurred in Uganda, with the 9-month Clinical Palliative Care Course. This course allowed clinicians to learn the basics of palliative care in 9 months, thereby greatly increasing access to care for their patients [39]. Educating the public is also crucial in order to debunk any myths and remove stigmas associated with cancer. This will allow earlier access to care and more effective treatments to take place. Local community leaders should be involved in advocacy and education campaigns. Cancer survivors should be highlighted so that the community can see cancer as a treatable diagnosis, instead of one that always leads to mortality.

Implementation of these recommendations can be challenging and requires coordination between healthcare providers, local leaders, and government figures. It also takes time to implement such changes, but even on a small scale, the benefits can be seen. Beginning with a few goal-oriented tasks (e.g., leading a community campaign to improve education about breast cancer on a local level and implementing policy changes to remove restrictions on morphine prescribing for cancer pain patients on a national level) can yield palpable results that can affect patient outcomes. Broader changes, such as strengthened tax capacity in order to improve healthcare budgets to build more facilities to improve access to care, certainly take more time but are part of the ultimate goal.

Conclusions

Low-resource settings provide unique challenges and opportunities in the treatment of cancer pain patients. One must start with the basics of assessment of cancer pain, including a detailed understanding of the etiologies of cancer pain. The types of cancer pain include somatic nociceptive, visceral nociceptive, and neuropathic pain. These can be affected by cancer disease states in a variety of ways including direct tumor-related pain, mass effect on structures adjacent to the tumor, destruction of tissues surrounding the tumor, and paraneoplastic pain syndromes. Treatment of cancer can also lead to certain pain states such as chemotherapy-induced peripheral neuropathy. Appropriate diagnosis leads to an effective treatment choice. The WHO ladder begins the first step of treatment with non-opioid pharmacologics and adjuvant medications, then moves on to the next step of opioid treatment for increasing or refractory pain. Interventional treatment modalities also play a critical role in effective cancer pain management. These may range from guided nerve blocks to implantable therapies such as intrathecal drug delivery.

The challenges presented in low-resource settings begin before the patient even presents. Prejudices against the healthcare system cause patients to avoid care or seek alternate care until their disease is at a more advanced state. This leads to a significantly more complex pain picture on initial presentation. Many treatment modalities that could have been utilized at an earlier state may now not be appropriate in such cases. Increasing public education in order to improve trust between the community and the healthcare system can ameliorate this issue. Healthcare providers must present a unified message with local, regional, and national leaders.

As the patient continues on his or her treatment journey, limited resources may adversely affect proper diagnosis and treatment. Modalities such as advanced imaging and modern therapies may not be available in low-resource settings, leading to poor treatment options for cancer patients. The opportunities here are more difficult to seize, as many of these issues are related to systemic issues such as lack of financial

support and governmental restrictions. Improvements to access of care will require longer-term solutions such as increased external aid and changes in local, regional, and national policies. This will allow improvement in healthcare infrastructure, which is necessary in order to provide better care to patients. Included in such improvements are a larger number of treatment centers, advanced imaging equipment, and better training for healthcare personnel. Medication availability to treat cancer pain is also highly variable in lowresource settings. This is linked to both financial support and strict laws in regard to opioids in certain countries. Unfortunately, this often leads to undertreatment of cancer pain as patients may not have access to the opioids they need; even if they do get access, it may be temporary as supplies are often inconsistent. There are many opportunities here to improve access to care, but again they require systemic changes such as improved funding and changes in the country's laws.

Many of the recommendations in this chapter take significant time, effort, and resources to implement. However, even small-scale changes can make significant differences in cancer pain patients' quality of life and outcomes. Lowresource settings provide settings in which the effects of these smaller-scale improvements can provide relatively significant effects.

References

- 1. Sheinfeld Gorin S, Krebs P, Badr H, Janke EA, Jim HS, Spring B, et al. Meta-analysis of psychosocial interventions to reduce pain in patients with cancer. J Clin Oncol. 2012;30(5):539–47.
- van den Beuken-van Everdingen MH, de Rijke JM, Kessels AG, Schouten HC, van Kleef M, Patijn J. Prevalence of pain in patients with cancer: a systematic review of the past 40 years. Ann Oncol. 2007;18(9):1437–49.
- Marmot S, Friel S, Bell R, Houweling TA, Taylor S. Closing the gap in a generation: health equity through action on the social determinants of health. Lancet. 2008;372(9650):1661–9. https:// doi.org/10.1016/S0140-6736(08)61690-6.
- Beiser M, Stewart M. Reducing health disparities: a priority for Canada (preface). Can J Public Health. 2005;96:S4–7.
- Santos Salas A, Fuentes Contreras J, Armijo-Olivo S, Saltaji H, Watanabe S, Chambers T, et al. Non-pharmacological cancer pain interventions in populations with social disparities: a systematic review and meta-analysis. Support Care Cancer. 2016;24(2):985–1000.
- Stanton EA. The human development index: a history. Working paper series no. 127. Political Economy Research Institute: Amherst; 2007.
- United Nations. Transforming our world: the 2030 agenda for sustainable development. A/RES/70/1. New York: UN Division for Sustainable Development Goals; 2015. https://sustainabledevelopment.un.org/post2015/transformingourworld. Accessed June 25 Jun 2020.
- Caraceni A. Evaluation and assessment of cancer pain and cancer pain treatment. Acta Anaesthesiol Scand. 2001;45(9):1067–75.
- 9. Beck SL, Falkson G. Prevalence and management of cancer pain in South Africa. Pain. 2001;94(1):75–84.

- Saini S, Bhatnagar S. Cancer pain management in developing countries. Indian J Palliat Care. 2016;22(4):373–7.
- Foley KM. Pain syndromes in patients with cancer. In: Bonica JJ, Ventafridda V, editors. Advances in pain research and therapy, vol. 2. New York: Raven Press; 1979. p. 59–75.
- Hjermstad MJ, Fainsinger R, Kaasa S, European Palliative Care Research Collaborative (EPCRC). Assessment and classification of cancer pain. Curr Opin Support Palliat Care. 2009;3(1):24–30.
- Caraceni A, Portenoy RK. An international survey of cancer pain characteristics and syndromes, IASP Task Force on Cancer Pain. International Association for the Study of Pain. Pain. 1999;82(3):263–74.
- Brant JM, Stringer LH. Cancer pain. In: Yarbro CH, Wujcik D, Gobel BH, editors. Cancer nursing: principles and practice. 8th ed. Burlington: Jones and Bartlett; 2018. p. 781–816.
- Fink RM, Gates RA, Jeffers KD. Pain assessment in the palliative care setting. In: Ferrell BR, Paice JA, editors. Oxford textbook of palliative nursing. 5th ed. Oxford, UK: Oxford University Press; 2019.
- 16. Subedi B, Grossberg GT. Phantom limb pain: mechanisms and treatment approaches. Pain Res Treat. 2011;2011:864605.
- 17. Portenoy RK. Treatment of cancer pain. Lancet. 2011;377(9784):2236–47.
- Dy SM, Asch SM, Naeim A, Sanati H, Walling A, Lorenz KA. Evidence-based standards for cancer pain management. J Clin Oncol. 2008;26(23):3879–85.
- Breivik H, Cherny N, Collett B, de Conno F, Filbet M, Foubert AJ, et al. Cancer-related pain: a pan-European survey of prevalence, treatment, and patient attitudes. Ann Oncol. 2009;20(8):1420–33.
- 20. National Cancer Policy Forum; Board on Health Care Services; Institute of Medicine; National Academies of Sciences, Engineering, and Medicine. Cancer care in low-resource areas: cancer prevention and early detection: workshop summary. Washington, DC: National Academies Press; 2016. https://www.ncbi.nlm.nih.gov/ books/NBK349625/. https://doi.org/10.17226/21892. Accessed 25 Jun 2020.
- Sun V, Borneman T, Piper B, Koczywas M, Ferrell B. Barriers to pain assessment and management in cancer survivorship. J Cancer Surviv. 2008;2(1):65–71.
- Farmer P, Frenk J, Knaul FM, Shulman LN, Alleyne G, Armstrong L, et al. Expansion of cancer care and control in countries of low and middle income: a call to action. Lancet. 2010;376(9747):1186– 93. https://doi.org/10.1016/S0140-6736(10)61152-X.
- Falkmer U, Järhult J, Wersäll P, Cavallin-Ståhl E. A systematic overview of radiation therapy effects in skeletal metastases. Acta Oncol. 2003;42(5–6):620–33.
- 24. Muayyad Ahmad M, Al Gamal E, Othman A, Nasrallah E. The National Survey. Knowledge, attitudes and practices towards cancer prevention and care in Jordan. Amman, Jordan: King Hussein Institute for Biotechnology and Cancer (KHIBC); 2011. p. 46.
- 25. McNicol E, Strassels SA, Goudas L, Lau J, Carr DB. NSAIDS or paracetamol, alone or combined with opioids, for cancer pain. Cochrane Database Syst Rev. 2005;1:CD005180. Update in: Cochrane Database Syst Rev. 2015;7:CD005180
- Humphreys K, Moos RH, Cohen C. Social and community resources and long-term recovery from treated and untreated alcoholism. J Stud Alcohol. 1997;58(3):231–8.

- Moore RA, Derry S, McQuay HJ. Cyclo-oxygenase-2 selective inhibitors and nonsteroidal anti-inflammatory drugs: balancing gastrointestinal and cardiovascular risk. BMC Musculoskelet Disord. 2007;8:73.
- 28. Chen YF, Jobanputra P, Barton P, Bryan S, Fry-Smith A, Harris G, Taylor RS. Cyclooxygenase-2 selective non-steroidal antiinflammatory drugs (etodolac, meloxicam, celecoxib, rofecoxib, etoricoxib, valdecoxib and lumiracoxib) for osteoarthritis and rheumatoid arthritis: a systematic review and economic evaluation. Health Technol Assess. 2008;12(11):1–278. iii
- 29. Porta-Sales J, Garzón-Rodríguez C, Llorens-Torromé S, Brunelli C, Pigni A, Caraceni A. Evidence on the analgesic role of bisphosphonates and denosumab in the treatment of pain due to bone metastases: a systematic review within the European Association for Palliative Care guidelines project. Palliat Med. 2017;31(1):5–25.
- Savage SR, Joranson DE, Covington EC, Schnoll SH, Heit HA, Gilson AM. Definitions related to the medical use of opioids: evolution towards universal agreement. J Pain Symptom Manag. 2003;26(1):655–67.
- 31. The British Pain Society. Faculty of Pain Medicine of the Royal College of Anaesthetists. Outcome measures. Jan 2019. https:// www.britishpainsociety.org/static/uploads/resources/files/ Outcome_Measures_January_2019.pdf. Accessed 26 Jun 2020.
- 32. Lötsch J, Rohrbacher M, Schmidt H, Doehring A, Brockmöller J, Geisslinger G. Can extremely low or high morphine formation from codeine be predicted prior to therapy initiation? Pain. 2009;144(1–2):119–24.
- Thompson PI, Joel SP, John L, Wedzicha JA, Maclean M, Slevin ML. Respiratory depression following morphine and morphine-6-glucuronide in normal subjects. Br J Clin Pharmacol. 1995;40(2):145–52.
- 34. Foley KM, Wagner JL, Joranson DE. Gelband. Pain control for people with cancer and AIDS. In: Jamison DT, Breman JG, Measham AR, Alleyne G, Claeson M, Evans DB, et al., editors. Disease control in developing countries. 2nd ed. Washington DC/New York: The International Bank for Reconstruction and Development/The World Bank/Oxford University Press; 2006. p. 981–93. https:// www.ncbi.nlm.nih.gov/books/NBK11800/.
- 35. Knaul FM, Farmer PE, Krakauer EL, De Lima L, Bhadelia A, Jiang Kwete X, Lancet Commission on Palliative Care and Pain Relief Study Group, et al. Alleviating the access abyss in palliative care and pain relief-an imperative of universal health coverage: the lancet commission report. Lancet. 2018;391(10128):1391–454. https://doi.org/10.1016/S0140-6736(17)32513-8.
- The World Bank. Population, total—Lower middle income 2018. https://data.worldbank.org. Accessed 25 Jun 2020.
- 37. Kalauokalani D, Franks P, Oliver JW, Meyers FJ, Kravitz RL. Can patient coaching reduce racial/ethnic disparities in cancer pain control? Secondary analysis of a randomized controlled trial. Pain Med. 2007;8(1):17–24.
- Joranson DE, Ryan KM. Ensuring opioid availability: methods and resources. J Pain Symptom Manag. 2007;33(5):527–32.
- Clark D, Wright M. The international observatory on end of life care: a global view of palliative care development. J Pain Symptom Manag. 2007;33(5):542–6.

Curriculum Development

Angela B. Creditt and Kevin Sing

In the United States, overall cancer incidence has been relatively stable over the last decade, ranging from 400 to 500 cases per 100,000 individuals [1]. Cancer death rates, however, have dropped by 29% from 1991 to 2017 as significant therapeutic advances have been achieved. Major developments in precision medicine, immunotherapies, and targeted therapies occur on a yearly basis, revolutionizing the delivery of oncologic care. For example, the 5-year overall survival of advanced melanoma in the interleukin era was 5–10%; with the combination of nivolumab and ipilimumab, that number has reached over 50% [2]. Thus, while disease-specific oncologic emergencies continue to occur, providers now contend with an array of new medications, as well as the life-threatening toxicities they can bring.

The development of new treatment strategies and a subsequently declining mortality rate has resulted in an increased number of visits to the emergency department (ED). In fact, most cancer patients will seek urgent care at least once during the course of their disease. Additionally, almost 60% of cancer patients who receive care in the ED will be admitted to the hospital [3]. Depending on hospital protocols and bed capacity, hospitalists may often assume the role of primary provider for cancer patients with hematology-oncologic physicians consulting to assist with management and treatment. Given the various specialties that will participate in the care of these patients, it is imperative that physicians receive thorough education on the identification and treatment of acute, and potentially life-threatening, oncologic morbidities.

A. B. Creditt (🖂)

K. Sing

Development of an Oncologic Emergency Medicine Curriculum

Medical students, residents, fellows, practicing physicians, and advanced practice providers - all have the potential to provide care to cancer patients, all can affect the outcome of these patients, and all will benefit from education on oncologic emergencies. However, each requires a different fund of knowledge and different educational approach. Undergraduate and postgraduate medical curricula must address core clinical competencies [4], but they are unique to each group of learners. For example, resident education is tailored toward milestones (on knowledge, clinical skills, attitudes, and other attributes) outlined by the Accreditation Council for Graduate Medical Education (ACGME) and the American Board of Emergency Medicine (ABEM) [5]. Here we will discuss and outline curriculum development as it pertains to an emergency medicine residency program. From this, adjustments can be made based on institutional differences or for other residency programs that will benefit from learning more about oncologic emergencies, such as internal medicine and family medicine. The second half of this chapter discusses special considerations for undergraduate medical education, hematology-oncologic fellowships, and continuing medical education.

Emergency Medicine Residency

Problem Identification and General Needs Assessment

Adult (age \geq 18 years) cancer-related visits to the ED have been increasing; there were 3.3 million visits in 2006 and 4.8 million in 2012. 4.2% of all adult ED visits in the United States were made by a patient with a cancer diagnosis, and this number is likely higher in facilities with an associated cancer treatment center. Furthermore, cancer-related ED visits are more common among older patients and are associated



JO

[©] Springer Nature Switzerland AG 2021 K. H. Todd et al. (eds.), *Oncologic Emergency Medicine*, https://doi.org/10.1007/978-3-030-67123-5_68

Department of Emergency Medicine, Virginia Commonwealth University Medical Center (VCU Health), Richmond, VA, USA e-mail: angela.creditt@vcuhealth.org

Division of Hematology, Oncology, and Palliative Care, Department of Medicine, Virginia Commonwealth University Medical Center (VCU Health), Richmond, VA, USA

with a significantly higher inpatient admission rate (59.7% compared to 16.3% of non-cancer-related visits) [6]. Given the prevalence of cancer-related visits and potential severity of illness, physicians must be prepared to provide high-quality care to these patients, including prompt identification and treatment of disease-specific and therapy-related oncologic emergencies.

Over their postgraduate medical training, emergency medicine residents face the challenging task of mastering a broad base of knowledge encompassing all medical disciplines with a focus on the acutely ill or injured and emergent resuscitation of critically sick patients. Under this umbrella falls oncologic emergencies. The specific topic of oncologic emergencies has few resources dedicated to offering curriculum direction or suggested programs that may guide educators on effective teaching methods. Yet, in 2016, the Model of the Clinical Practice of Emergency Medicine (EM Model) added "oncologic emergencies" as a required curricular component requiring residents and practicing emergency physicians to obtain core fundamental knowledge on this subject in order to provide proper treatment and patient care [7]. What the EM Model did not do is provide education objectives or an outline of what content should be covered in the curriculum.

Beyond the EM Model, the ACGME, the body responsible for accrediting all graduate medical training programs, publishes "common program requirements" for each residency and fellowship. Essentially, these requirements are basic standards for the knowledge, skills, and attitudes deemed necessary that residents should acquire to appropriately take care of patients [8]. From this, each specialty has its own document outlining program requirements specific to that discipline. For emergency medicine, it states:

The curriculum must contain the following educational components: a set of program aims consistent with the Sponsoring Institution's mission, the needs of the community it serves, and the desired distinctive capabilities of its graduates...competencybased goals and objectives for each educational experience designed to promote progress on a trajectory to autonomous practice...a broad range of structured didactic activities.

Additionally, it describes the competencies and milestones, such as professionalism, patient care and procedural skills, medical knowledge, etc. that educators must use to track and evaluate individual residents [9]. While the ACGME program requirements provides a comprehensive curricular foundation, as does the EM Model, it does not provide specific educational objectives or itemized curricular components that should be taught to each learner.

To supplement this, emergency medicine residency programs frequently utilize either *Tintinalli's Emergency Medicine: A Comprehensive Study Guide*, 9th edition (McGraw-Hill Education/Medical; 2019), or *Rosen's* Table 68.1 Topics included in emergency medicine textbooks

Textbook and chapter	Conditions discussed
Tintinalli's emergency	Airway obstruction
medicine: A comprehensive	Bone metastases and pathologic
study guide	fractures
Chap. 239, emergency	Spinal cord compression
complications of malignancy	Malignant pericardial effusion
	Superior vena cava syndrome
	Hypercalcemia of malignancy
	Hyponatremia due to syndrome of
	inappropriate anti-diuretic hormone
	(SIADH)
	Adrenal crisis
	Tumor lysis syndrome
	Febrile neutropenia
	Hyperviscosity syndrome
	Thromboembolism
	Nausea and vomiting
	Extravasation of chemotherapeutic
	agents
Tintinalli's emergency	Emergencies related to local tumor
medicine: A comprehensive	effects
study guide	Emergencies related to biochemical
Chap. 240, Emergency	derangement
Complications of	Emergencies related to hematologic
Malignancy	derangement
	Emergencies related to therapy
Rosen's emergency	Febrile neutropenia
medicine: Concepts and	Metastatic spinal cord compression
clinical practice	Malignant pericardial disease
Chap. 115, Selected	Hypercalcemia
Oncologic Emergencies	Tumor lysis syndrome
	Leukostasis
	Superior vena cava syndrome

Emergency Medicine: Concepts and Clinical Practice, 9th edition (Elsevier; 2018), as a reference or guideline for educational components. Each textbook includes a section on oncologic emergencies; see Table 68.1 for specific topics covered. Both resources include content on the more commonly known oncologic emergencies, while *Tintinalli's* seems to have a more diverse list of topics. Neither currently discusses newly identified cancer-related emergencies such as immunotherapy toxicities.

Another core component of emergency medicine in general but also particularly as it relates to patients with cancer is palliative and end-of-life care. The 2016 EM Model notes that clinical knowledge in healthcare coordination, including end-of-life and palliative care, advanced directives, coordination with hospice, and organ donation, is integral to the practice of emergency medicine [7]. Yet, there is a perceived deficit in knowledge of palliative and end-of-life care by emergency residents and physicians [10], and only 59% of residencies have been found to teach these competencies in their program [11]. In part, this could be due to barriers within the ED such as perceived lack of time, lack of experience or expertise among faculty, and lack of faculty interest in palliative care. Additionally, the majority of programs have access to hospice and palliative medicine consultants, thereby potentially negating opportunities for clinical education [11].

In 2007, a formal curriculum entitled The Education in Palliative and End-of-Life Care for Emergency Medicine (EPEC-EM) was created with the support of the National Institutes of Health. The EPEC-EM was developed by emergency physicians and nurse educators to enhance education in these essential clinical competencies as it specifically relates to the field of emergency medicine. However, despite this initiative, there remains a general lack of formal palliative care training in emergency medicine residency programs across the country. Emergency residents and physicians feel palliative care skills are important and desire more dedicated training in this field [11].

Given the lack of a generalizable detailed emergency medicine curriculum and related literature, it is difficult to define precisely the didactic education residents currently receive on oncologic emergencies. Furthermore, clinical exposure likely varies considerably based on characteristics of the residency program. For example, those who train at an institution with an affiliated cancer treatment center likely see a significantly higher cancer patient population than those without a center. This will ultimately translate to improved clinical experience and knowledge, confidence in workup and diagnosis, and an enhanced ability to manage emergent conditions affecting cancer patients, thereby resulting in higher-quality care and improved patient outcomes.

Goals and Objectives

Based on the problem identified and needs assessment outlined above, the overall goal and specific objectives for an oncologic emergency medicine curriculum can be developed. These objectives will help establish curricular content and learning approaches, help focus the learner, and provide a foundation for evaluation. The overall goal for a curriculum dedicated to oncologic emergencies is to expand and develop resident physician knowledge, clinical skills, and behaviors to effectively diagnose and treat patients who present with an oncologic emergency, thereby improving medical decision making and patient clinical outcomes. Specific objectives include the following:

- Identify and manage emergent medical conditions that occur in cancer patients, including local tumor effects, complications related to cancer treatment, hematologic derangements, and biochemical abnormalities.
- Understand novel cancer therapies, associated side effects, management, and treatment options.
- Exhibit proficient clinical skills through consideration of relevant differential diagnoses, workup, and management

Table	68.2	Suggested	oncologic	emergency	medicine	educational
topics						

Local tumor effects	Airway hemorrhage, airway obstruction, bone metastases and pathologic fractures, brain metastases, malignant pericardial effusion, malignant pleural effusion, malignant spinal cord compression, superior vena cava syndrome
Complications related to cancer treatment	Bone marrow transplant complications, chemotherapy adverse effects, extravasation of chemotherapeutic agents, immunotherapy and CAR-T cell toxicities, radiation therapy complications, symptom management (i.e., nausea and vomiting, pain control)
Hematologic derangements	Blast crisis, hyperviscosity syndrome, leukostasis, neutropenic fever, pulmonary embolism
Biochemical abnormalities	Adrenal insufficiency, hypercalcemia, hyponatremia secondary to syndrome of inappropriate anti-diuretic hormone (SIADH), tumor lysis syndrome

based on a cancer patient's medical history and clinical presentation.

- Recognize the following conditions: neutropenic fever, superior vena cava syndrome, tumor lysis syndrome, malignant spinal cord compression, blast crisis, immunotherapy and CAR-T cell toxicities, hypercalcemia, pulmonary embolism, hyperviscosity syndrome, malignant pericardial effusion, bone marrow transplant complications, and radiation therapy complications. See Table 68.2 for a more conclusive list of conditions.
- Learn supportive treatment plans for patients who need symptomatic care (i.e., nausea, vomiting, cancer-related pain, etc.) and understand when to include consultants such as palliative medicine and hematology-oncologic specialists to optimize patient management outside of the ED.
- Demonstrate sensitivity and commitment to ethical principles by assessing goals of care, including advanced care directives, and the patient's emotional state during their ED visit.
- Understand appropriate disposition including: discharge home, admission to the hospital or hospice, and transfer to higher level of care.

Educational Strategies

While every resident has gained a foundation of knowledge in oncologic from medical school, their awareness and understanding of cancer-related emergent conditions may vary considerably (see the section on medical school education later in this chapter). Therefore, to achieve curriculum objectives, appropriate educational content and teaching methods should be selected. Learners can obtain lower-level knowledge through formal lectures, suggested reading assignments, and asynchronous online modules. Case-based problem-solving exercises and other small group modules, such as simulation, that rely on active participation by learners can then be implemented to enhance clinical reasoning skills. Finally, bedside teaching and on-shift clinical education must occur to cement knowledge learned through didactic instruction.

Ideally, the majority of education will occur during weekly resident conferences in a set educational block dedicated to oncologic-related topics in emergency medicine. The amount of time allotted to this block is dependent on several factors, such as 3- versus 4-year residency programs and how often the didactic curriculum is repeated during a resident's postgraduate education. In general, a minimum of 8 to 10 hours of fundamental oncologic emergency training using variable teaching methods is necessary to enhance knowledge retention. These methods can (and should) include PowerPoint® (Microsoft, Redmond, WA) lectures, small group education including problem- or case-based and team-based learning, simulation and role playing, flipped classroom, and bedside teaching.

Utilizing PowerPoint lectures for a portion of teaching in oncologic emergencies allows for the delivery of concrete, structured information to a large group of people. This provides a knowledge base to residents than can then be augmented using other learning techniques. In particular, small group learning creates high-yield and interactive didactic sessions that yield improved learner engagement, teamwork, clarification of knowledge gaps, and opportunities for resident assessment. Ideally, the small group session would be resident directed with a faculty leader available to answer questions, clarify any points of confusion, provide additional information as needed, and keep the group focused and on track [12].

Small group sessions can be conducted using team-based learning, case-based learning, or simulation/role play. In team-based learning, residents first complete an educational assignment. Subsequently, in conference they are tested on pre-class material via an individual readiness assurance test (iRAT), then a group readiness assurance test (gRAT), followed by a group exercise challenging residents to apply their knowledge to a scenario as a team. A single instructor is required to review learning points and clarify any confusion. Team-based learning has been shown to improve learning outcomes and examination scores as well as communication and teamwork skills [12]. Case-based learning utilizes vignettes of real or hypothetical patients to facilitate a discussion at each decision point. Throughout the case, an instructor asks questions to highlight different learning objectives, such as clinical presentation, development of a differential diagnosis, treatment, etc. This technique can also be used to tackle more challenging objectives within the oncologic emergency medicine curriculum, such as communication with a patient and/or their family or discussions on goals of care and do-not-resuscitate orders [13]. See Table 68.3 for an example case that can be used for case-based small group learning.

Table 68.3 Example case for small group learning

Case presentation	
History of presenting illness	A 67-year-old male with a history of hypertension, hyperlipidemia, coronary artery disease, and prostate cancer (last chemotherapy was 6 month ago) presents to the ED with back pain that began 6 days ago. Reportedly his pain extends from his mid- to lower back, is described as aching and sharp with occasional radiation into his legs bilaterally, and associated with intermittent tingling in his right foot. He says he thinks he may have injured himself while walking his dog but denies fall or specific injury. Additionally, he endorses back pain a few years ago but is unsure if his pain today feels similar. Patient denies fever, chills, bowel or bladder dysfunction, saddle anesthesia, weakness, numbness, history of intravenous drug abuse, or other symptoms
Medical history	Hypertension, hyperlipidemia, coronary artery disease, prostate cancer
Surgical history	Denies
Social history	Drinks alcohol socially, denies tobacco and illicit drug use
Physical exam Stop: Ask the followin	Vital signs: BP 156/95 HR 87 RR 98% SpO2 98% on RA T 37.0 General: Awake and alert, mildly cachectic male in no acute distress Skin: Clean, dry, intact, no rash Head: Atraumatic, normocephalic Eyes, ears, nose, throat: Pupils are equal round and reactive to light Neck: Supple, full range of motion, trachea is midline Cardiovascular: Regular rate and rhythm, no murmur, no edema Respiratory: Clear to auscultation bilaterally, non-labored Abdomen: Soft, nontender, nondistended Back: Limited range of motion secondary to pain, tenderness to palpation in the lumbar and thoracic spine Musculoskeletal: Strength is 5/5 to bilateral lower and upper extremities with full range of motion throughout Neurologic: Alert and oriented, sensation is within normal limits, difficulty ambulating secondary to pain without ataxia, deep tendon reflexes 2/4 bilaterally, normal Babinski reflex g questions
What is your	g questions
what is your differential diagnosis?	Spinal stenosis Herniated disc Musculoskeletal strain Bone metastasis

What tests would you order for this patient?	Complete blood count Comprehensive metabolic panel Magnetic resonance imaging (MRI) of the cervical, thoracic, and lumbar spine with and without contrast Or computed tomography (CT) scan of the thoracic and lumbar spine (if not at
	MRI-capable facility)
At this time, provide lear requested for analysis	arners with the lab work and imaging
What is the diagnosis?	Malignant spinal cord compression
<i>How would you manage this patient?</i>	Administer intravenous dexamethasone 10 mg STAT Consult neurosurgery to evaluate for possible operative intervention Consult radiation-oncologic for emergent radiation therapy Provide pain control Admit to hematology-oncologic or medicine service (if not an operative candidate)

Perform case debrief

What went well? What did not? Are there any questions pertaining to this case? Has anyone taken care of a patient with malignant spinal cord compression? If so, what were barriers to care? Any general comments or concerns?

Briefly discuss malignant spinal cord compression

Include pathophysiology, clinical presentation, diagnosis, and intuition-specific management as well as management pathways based on other clinical scenarios (i.e., at rural ED)

Simulation, on the other hand, provides more realistic, high-fidelity learning using mannequin simulators or standardized patients. Simulation delivers engaging, team-based active learning that can expose residents to more critical patient scenarios and has been shown to improve comfort in performing procedures, clinical performance, and knowledge retention [12]. Neutropenic fever with sepsis, malignant pericardial effusion with tamponade, spinal cord compression with progressive neurologic deficit, and airway obstruction secondary to a mass are scenarios that are particularly suitable for simulation.

A novel approach to resident education and small group learning known as the "flipped classroom" is also a valuable method for teaching oncologic emergency medicine. With this method, traditional in-class lectures and "homework" are reversed in that learners are assigned content to review at home prior to weekly didactic conference. This homework might include completing an online module, watching a video-recorded lecture, listening to a podcast, or reading a journal article or assigned textbook section to facilitate meeting an educational objective. In conference, instead of a traditional PowerPoint lecture, learners are asked to apply their new knowledge through small group activities requiring them to work through a problem or patient case, with the assistance of a skilled group leader. When successful, this technique creates a framework of core knowledge that is reinforced and cemented through an interactive process [14].

Lastly, longitudinal education in oncologic emergencies throughout the year is helpful to cement knowledge gained through patient encounters and didactic education. This can be done with on-shift bedside teaching, oral board cases, and teaching rounds. Some institutions utilize group sign-out or have a specified time dedicated to a short period of education on a selected topic. This topic could be something selected previously by the scheduled instructor or it could be based on a case encountered during the shift. Additionally, sending a podcast, new journal article, or other nugget of information to residents via email can also enhance their education.

Considering the variable teaching methods outlined above, a curriculum in oncologic emergency medicine that is learner-centered, facilitates engagement, and optimizes knowledge retention can be easily developed. See Table 68.4 for an example of a 10-hour curriculum.

 Table
 68.4
 Example
 of
 an
 oncologic
 emergency
 medicine

 curriculum

 </td

0 1 1		
Oncologic emergence	cy medicine didactic block	
Day 1		
Subject	Teaching method	Time
Introduction to oncologic emergency medicine	Flipped classroom Pre-conference assignment: Sadik, et al., "attributes of cancer patients admitted to the ED in one year" [3] and/or Rivera, et al., "trends in adult cancer-related emergency department utilization" [6] Alternative method: PowerPoint lecture	30 min
Hematologic derangements and cancer	PowerPoint lecture Including neutropenic fever, blast crisis, hyperviscosity syndrome, leukostasis. Alternative method: Flipped classroom, team-based learning	40 min
Biochemical abnormalities and cancer	PowerPoint lecture Including hypercalcemia, tumor lysis syndrome, hyponatremia (syndrome of inappropriate anti-diuretic hormone), adrenal insufficiency. Alternative method: Flipped classroom, team-based learning	40 min
Break (10 min)		
Immunotherapy and its toxicities	Flipped classroom Identification of immunotherapy agents and CAR-T cell therapies. Toxicities related to these novel agents and potential treatments. Alternative method: PowerPoint lecture, team-based learning	30 min

(continued)

Oncologic emergence	cy medicine didactic block	
Cancer prevention and diagnosis in the ED	PowerPoint lecture and team-based learning Guest from hematology-oncologic to discuss the role of the ED in prevention and diagnosis of management. Alternative method: Flipped classroom	30 min
Neutropenic fever Immunotherapy toxicity Biochemical abnormalities and cancer Difficult communication	 Small group: Mix of case-based, team-based learning, and role play Residents will be divided into four small groups, each group will rotate through each small group session. Difficult communication (and breaking bad news) will be done via role play. Alternative method: Simulation, flipped classroom, oral boards, role play 	2 hours 30 min at each station
Day 2		
Subject	Teaching method	Time
Topics in palliative care	PowerPoint lecture and group discussion Guest from palliative medicine to review relevant topics including breaking bad news, end-of-life care in the ED, goals of care discussions, etc. Alternative method: Flipped classroom	1 hour
Supportive care in the ED	PowerPoint lecture Outline treatment plans for symptomatic care (i.e., nausea and vomiting). Discuss pain control as it relates to the cancer patient – This may include palliative medicine guest and/or clinical pharmacist. Alternative method: Flipped classroom, team-based learning	30 min
Emergencies related to local tumor effects	Flipped classroom Including superior vena cava syndrome, malignant airway obstruction, spinal cord compression, brain metastasis, malignant pericardial effusion and tamponade, acute airway hemorrhage. Alternative method: PowerPoint lecture, team-based learning, case-based learning	50 min
Break (10 min)		
Chemotherapy, radiation, and associated toxicities	PowerPoint lecture Including extravasation of chemotherapy agents, medication- specific side effects, radiation complications, bone marrow transplant complications. Alternative method: Flipped classroom, team-based learning	30 min
	-	

Table 68.4 (continued)

Oncologic emergency medicine didactic block		
Airway obstruction	High-fidelity simulation modules	2 hours
Spinal cord	Residents will be divided into four	30 min
compression	small groups, each group will	at each
Malignant	rotate through the four different	station
tamponade	simulation sessions.	
Topics in palliative	Mannequins for difficult	
care	intubation and pericardiocentesis	
	are needed.	
	Alternative method: Case-based	
	learning, team-based learning, oral	
	boards	

Implementation

For an oncologic emergency medicine curriculum to be successful, attention must be given to factors important to implementation success. The curriculum developer must ensure that sufficient political and financial support, resources, and administrative structures are in place [4].

First, support should be obtained from curriculum stakeholders. This will include the residency program director and the faculty leader of resident didactic curriculum, followed by the program's core faculty members. Because this curriculum will require preparation and teaching from those responsible for residency education, it is important that core faculty members support the initiative. Additionally, it would likely be beneficial to discuss the curriculum with the hematology-oncologic chair, fellowship program director, or other departmental leaders to ensure that the educational objectives and overall curriculum not only meet the educational needs of the residency, but also properly support the needs of the institution and local cancer population. Furthermore, support should be gained from learners. Adult learners need to understand why learning about oncologic emergencies is important. Having support from learners will foster successful curriculum implementation, particularly when independent learning is required, such as with flipped classroom modules.

Next, adequate resources should be identified. Resources will typically include personnel, facilities, time, and funding (if needed).

Curriculum developer/director, expert faculty educators
to teach didactics and lead small group activities,
simulation director to coordinate simulation sessions, and
administrative staff to communicate schedules and
collect evaluation reports are all necessary for an
oncologic emergency medicine curriculum. If faculty are
not experienced with oncologic emergencies, consider
faculty development sessions to facilitate enhanced
knowledge and skill (see "Continuing Medical
Education" further in this chapter).

Facilities	Resident conference room for didactic lectures, small group discussions, and flipped classroom activities. Simulation center for role playing and case-based learning. Clinical site (i.e., ED) to provide an appropriate volume and variable patient presentation to enhance clinical skills. Equipment, including a computer with projector, iPAD for simulation modules, simulation mannequin, software for learning modules, etc.
Time	Curriculum director will need time to coordinate curricular details, analyze evaluations, and provide feedback. Learners require time for independent learning, to attend educational sessions, and to provide feedback. Faculty need time to prepare and teach. Given that oncologic emergencies will be part of an existing resident didactic curriculum, this time is likely already part of core faculty protected hours and professional obligations.
Funding	Depending on the institution, funding may be required to purchase necessary equipment or software to support curriculum implementation. However, intuitions may already have budgets allocating resources for these items or necessary facilities/equipment may already exist and simply need redeployed to accommodate educational activities.

Before beginning a new oncologic emergencies curriculum (or making modifications to an old curriculum), it is advantageous to anticipate and address potential barriers that may occur. Barriers may relate to people, time, education strategies, finances, or other resources [4].

Educators	Competing demands from other job responsibilities or teaching obligations may limit faculty involvement. Limited knowledge of oncologic emergency medicine due to rapidly evolving treatment strategies may require that educators participate in developmental sessions or independent continuing education endeavors.
Learners	Residents may not understand the importance of each educational objective and therefore not actively participate in didactic education. A successful curriculum implementation necessitates comprehensive knowledge retention and achievement of clinical skills and therefore requires that residents take ownership of their own learning by investing time in studying oncologic emergencies outside scheduled conferences.
Time	Providing comprehensive didactic education in only 5 hours a week during residency is a challenging task. Additionally, residents are not always available to attend conference due to variable clinical rotation requirements, duty hour restrictions, or dispersal to different clinical sites.
Education strategies	When using a flipped classroom teaching strategy, if learners do not complete the pre-conference assignment, it limits the likelihood of a successful in-conference session. In addition, quieter residents or those who do not like to speak in groups may not benefit from in-conference discussion [14].
Funding or other resources	There may not be funding to support a curriculum developer/director or to purchase equipment or software that will enhance resident education.

Once support has been gained, resources allocated, and barriers addressed, it is time for curriculum implementation. Typically, this type of curriculum will not need a formal pilot period prior to full implementation. However, it may be worthwhile to pilot simulation and case-based learning sessions with fellow faculty members or residents to ensure they are designed appropriately to achieve educational objectives and knowledge retention. If flaws are discovered, adjustments can be made accordingly. Once piloting is complete, the new or improved oncologic emergency medicine curriculum should be scheduled into the overall resident didactic program.

Following implementation, a plan should exist for curriculum maintenance and enhancement based on feedback and evaluation. A curriculum developer may consider the first year of implementation a "pilot" cycle during which time achievement of goals and objectives is assessed, feedback from residents and faculty obtained and contribution to milestones analyzed. This information can then be used to refine the curriculum for subsequent cycles with a strategy in place for continuous improvement [4].

Evaluation and Feedback

Importantly, consideration should be given to maintaining and enhancing the oncologic emergency medicine curriculum over time. Evaluation helps judge whether goals and objectives were met, provides information that can be used to improve or alter the curriculum, allows for the assessment of resident achievement, and can serve as data for possible publication. Evaluation methods and questions should be developed based on the desired outcome and information sought [4].

The development of feedback should be methodical. First, the curriculum director should identify users of the evaluation, including residents, faculty instructors, program director, and the curriculum developer. Depending on the institution, this may also include the department chair, a hematology-oncologic representative, and individuals or organizations providing financial contributions. Additionally, consider how the evaluation will be used [4]. For the oncologic emergencies curriculum, this will include performance assessment of both the learners (i.e., residents) and the entire program.

Users	Residents will use evaluation and feedback information to
	understand and improve their own performance and clinical
	skills.
	Faculty instructors/small group leaders will use the data to
	enhance their teaching methods and delivery, thereby
	improving learner satisfaction.
	Curriculum developer/director will use the data to discover
	strengths and weaknesses of the program resulting in
	improved educational strategies.
	Program leadership will use resident evaluations and
	feedback to assist with assessment of achievement of clinical
	competencies and milestones.

Availability of resources will dictate evaluation questions and methodology [4]. For the purposes of an oncologic emergency medicine curriculum, necessary resources are not extensive, minimizing a potential barrier in obtaining feedback.

Resources	An online platform is needed to serve as a channel for
Resources	All olimic platorin is needed to serve as a challer for
	residents and other individuals to complete evaluations.
	This can be done through residency management
	software (e.g., New Innovations®, Uniontown, OH).
	Administrative support staff or the curriculum director is
	needed to assist with evaluation requests and data
	analysis.

Evaluation questions should be congruent with curriculum objectives and relate to a specific learner, process, or clinical outcome [4]. During the first year of curriculum implementation, evaluations should be completed at the end of a pre-conference assignment, following a lecture, small group learning exercise, or simulation session, and after the curriculum is complete. This will allow the developer to assess the effectiveness of the curriculum and make necessary changes. In the subsequent years, evaluations can be tailored more toward the instructor, teaching method, and learner.

Questions will be answered as strongly agree,
agree, neutral, disagree, strongly disagree, and not applicable:
The resident appeared to complete pre-
conference assignments.
The resident was engaged and paid attention to
The resident performed adequate history and
physical exam.
The resident developed an appropriate
differential diagnosis and management plan.
The resident demonstrated clinical reasoning
skills during case-based learning activities.
The resident actively participated in small group activities.
The resident communicated effectively with other residents in their group.
The resident demonstrated professionalism and
respectability toward peers and faculty
instructors.
Questions will be answered as strongly agree, agree, neutral, disagree, strongly disagree, and not applicable:
I was impressed by the faculty member's overall
knowledge of oncologic emergencies.
The content presented is relative to my education
and practice.
The presentation was evidence based and without
bias.
The faculty member was an effective teacher
with clear educational objectives.
The faculty member communicated effectively with residents during small group education.

Curriculum	What do residents identify as major strengths and
evaluation	weaknesses of the oncologic emergencies
questions	curriculum? How could the curriculum be
	improved?
	What was the perceived effectiveness of the
	PowerPoint lectures, small group learning,
	simulation sessions, pre-conference assignments,
	and flipped classroom modules in providing
	valuable education on oncologic emergencies?

Finally, adequate data should be collected to ensure useful analysis. Failure to collect important information or low response rates can compromise the value of an evaluation. Data collection, therefore, should be designed to maximize response rates, practicality, and efficiency of evaluation completion. Following collection, data must be analyzed and reported. When considering how to report assessment information, it is helpful to consider the needs of users. Specifically, residents will benefit from immediate feedback to allow for the information to be processed while the experience is still fresh in their mind [4].

Data	Electronic collection of individual and curriculum
collection	assessment should be utilized. This can be implemented
	within an already existing generic evaluation method or
	a more specific approach can be developed. Each
	component as well as the overall oncologic emergencies
	curriculum should be evaluated separately.
	During the first year of implementation, it is
	advantageous to obtain specific data on instructional
	strategies, teaching methods, individual instructors, and
	overall program content, to maximize curricular
	enhancement and address unforeseen issues.
Data	Analysis of completed evaluations should occur in real
analysis	time to optimally facilitate necessary changes while the
	memory of curriculum details and experience are fresh.
Reporting	Immediate formative feedback should be given to
of results	residents following any small group activity involving
	direct observation by a faculty member. If significant
	concerns are appreciated, this information should be
	given to the residency program director.
	Reported feedback should be constructive, using
	succinct, clear language.
	A summary of formative and summative information
	must be given to the curriculum director so that
	improvements can be made and unmet curricular needs
	assessed.

While the general process of evaluation and feedback can be cumbersome, it is essential to assess resident learner performance and to guide effective education.

Medical School Education in Oncologic

Acquiring knowledge and clinical skills that will enable physicians to optimally care for cancer-related emergencies begins with undergraduate medical education. Medical school is essential to provide a foundation of cancer knowledge that can be expanded upon during residency. However, delivering a comprehensive oncologic curriculum to medical students is challenging due to the field's multidisciplinary nature and lack of data on effective teaching approaches [15, 16]. To better characterize oncologic education, surveys of medical students across the United States suggest the following: students are significantly more confident in the basic science of cancer than in workup/diagnosis, treatment, and interacting with oncologists; pre-clinical oncologic education is often fragmented and varies considerably among intuitions [16]; and few schools have mandatory clinical clerkships dedicated to oncologic [15, 16]. Cancer was also found to be under-emphasized in medical school curriculums [15]; it is the second leading cause of death in the United States (behind heart disease) [17], yet it reportedly receives the fourth most curricular time among the six most common causes of death [15].

Another component introduced during medical school and relevant to caring for patients with cancer-related emergencies is palliative and end-of-life care. Palliative care can (and should) include patient and family communication to discuss goals of care, pain and symptom management for critically ill and dying patients, coordination of care, and optimizing quality of life. US medical schools are not currently required by the Liaison Committee on Medical Education (LCME) to teach palliative care, though it does necessitate that curricula include instruction on end-of-life care. The Association of American Medical Colleges (AAMC), however, does recommend that undergraduate medical students learn about shared decision making, communication of bad news, advanced directives, end-of-life wishes, do-not-resuscitate orders, and palliative care [18].

Surveys collected every 5 years since 1975 find dramatic variability among medical schools on the amount of time spent teaching end-of-life care, ranging from 2 to 80 hours over 4 years. Less than one-third reported a clerkship dedicated to end-of-life care, and of these, only one-half are required. All but one responding medical school provided dedicated palliative care education. Another survey reports that of 47 responding medical schools, 30% had a required course, 19% had a required rotation, 15% offered an elective, and 7% offered no course or rotation. The remaining 29% of schools integrated palliative care teaching within another mandatory rotation. Fourth-year students receiving formal training in palliative care reported superior competence and knowledge in caring for the critically ill and dying [18].

This analysis of undergraduate medical education in oncologic and palliative care indicates that medical schools can benefit from a more defined curriculum. With enhanced cancer survivorship, the need for cancer-related knowledge and skills among generalists and specialists will increase proportionally [16]. However, given the limited amount of didactic time available for any one area of medicine, developing an effective oncologic curriculum is challenging. Neeley et al. suggest the following [15]:

- Improve the organizational structure and coordination of oncologic didactics across all years of medical school to avoid redundancy or accidental omission of content and to ensure there is balance among the types of cancer covered and types of educators who are teaching.
- 2. Request multidisciplinary contribution to curriculum development to facilitate a more inclusive approach to teaching cancer management.
- 3. Avoid pure disease-site and organ-system-specific courses; instead supplement them with education on general principles of oncologic, survivorship care, palliative medicine, and the role of primary care physicians in cancer prevention, diagnosis, and management.
- 4. Create a multidisciplinary oncologic clerkship that provides broad exposure to caring for patients with cancer and incorporates outpatient oncologic exposure for students who plan to pursue non-oncologic fields of medicine.
- 5. Support the development of national guidelines for medical school education in oncologic, as has been established in Europe, Canada, and Australia.

Another consideration is to create a course to supplement oncologic education similar to the one developed by the European School of Oncologic (ESO). This elective 5-day training program is dedicated to improving medical students' oncologic knowledge and clinical skills. The program includes education in prevention, epidemiology, clinical presentation, diagnosis, and multimodal therapeutic management. It also includes sessions on oncologic emergencies and palliative care [19].

Finally, optimal medical school education in oncologic must include palliative medicine training. According to Horowitz et al. and Head et al., the following must be considered for our undergraduate medical education: 1) palliative care education cannot be limited to electives; it should be integrated into existing courses, such as the practice of clinical medicine, ethics, and pharmacology, as well as in relevant clinical clerkships; 2) basic principal palliative care competency requirements should be established and implemented in all medical schools; and 3) palliative care training should provide developmentally appropriate education for students to achieve these competencies at each stage of schooling [18, 20].

Hematology-Oncologic Fellowship Training

While many cancer patients with concerning symptoms seek treatment at urgent care centers or emergency departments, they may also present to their oncologic clinic or call an oncologic provider after hours. Oncologists also routinely manage these patients after hospital admission. It is thus vital for hematologists and oncologists to be expert in the diagnosis and management of oncologic emergencies.

Though some experience is gained during internal medicine residency, fellows require more advanced and specific knowledge when they begin treating cancer patients early in their fellowship. Currently, hematology-oncologic fellowships in the United States either disperse inpatient clinical duties (including consult and bone marrow transplant services) throughout all 3 years or concentrate them into a more intense first year of fellowship. Didactics and continuity clinics supplement this training. While these experiences provide fellows with gradual hands-on knowledge and understanding of the diagnosis and management of oncologic emergencies, the establishment of a dedicated educational module early in fellowship would enhance learners' confidence in clinical care.

Oncologic emergencies are broad and encompass every system of the human body. For disease-specific emergencies, they can be categorized and taught via systems [21]. Systems categories include metabolic and endocrine (e.g., hypercalcemia, tumor lysis, adrenal), hematologic (e.g., leukostasis, hyperviscosity), neurologic (e.g., vasogenic edema from brain metastases, malignant spinal cord compression, paraneoplastic syndromes such as Lambert-Eaton), cardiovascular (e.g., pericardial effusion, superior vena cava syndrome), and pulmonary (e.g., pleural effusion, airway obstruction, and hemorrhage). The other major type of oncologic emergencies is treatment-related. This subject is particularly important to hematology-oncologic fellows because their exposure from residency didactics and clinical experience is limited. Some examples include immunotherapy-related adverse events, coronary effects from fluoropyrimidines, arrhythmias from ibrutinib, pulmonary toxicity from bleomycin, thrombotic microangiopathy from gemcitabine, and cytokine release syndrome from CAR-T or blinatumomab.

Ideally, a module in oncologic emergencies would include short, case-based didactic sessions at the beginning of fellowship. Cased-based learning has been shown to promote effective learning through increased engagement and authentic clinical practice scenarios [22]. In case-based learning, specific scenarios that resemble realistic patients are created for learners to solve and generate discussion regarding management and treatment plans. This allows fellows to connect theory to a practice situation, thereby integrating basic science and clinical management resulting in active learning and enhanced retention of knowledge [22]. For example, oncologic pharmacists can be incorporated into case-based learning sessions to provide training on specific chemotherapies and medication toxicities to which most fellows likely would not have had extensive exposure. Educational modules on treatment-specific chemotherapy and other pharmaceutical toxicities can then be incorporated into lectures and clinical discussions spread throughout the year. The goal of these sessions would be to reinforce concepts and management strategies to broaden clinical knowledge and skills throughout fellowship.

Including such a module would be a relatively small change that can be made to most hematology-oncologic fellowships without sacrificing the overall curriculum or other valuable learning concepts. Then, instead of progressive learning over several years, fellows will achieve a better knowledge base earlier in fellowship resulting in enhanced quality of care for cancer patients with emergent conditions. To facilitate widespread implementation of this module, the American Society of Hematology (ASH) and American Society of Clinical Oncologic (ASCO) may be able to help. The ASH and ASCO work with the ACGME to create and enforce program requirements and milestones; they also provide workshops and various online programs for both program directors and fellows. These societies could create a recommended oncologic emergencies module that would enable individual programs to then implement this didactic program in a way that suits their specific educational needs with regard to clinical environment, time, and staffing requirements.

Feedback is fundamental to this process. Once the introductory oncologic emergencies module is complete, learners should be asked to provide feedback and evaluate the teaching methods and educational concepts, and instructors to promote effective change. This should be repeated at the halfway point and the end of the first year of fellowship to gather additional comments or critiques as fellows gain further knowledge and clinical experience. Participation should be confidential so opinions can be voiced freely. Improvements and changes can then be made for the new fellows and the entire educational process can be continually refined.

Oncologic Emergency Medicine Fellowships

As the cancer population grows, more and more will seek emergent care. Understanding this need, two oncologic emergency medicine fellowships have been developed, one at The University of Texas MD Anderson Cancer Center and another at The Ohio State University [23].

The University of Texas MD Anderson Cancer Center opened an academic ED within its comprehensive cancer center in 2010. This 40-bed center provides care to almost 26,000 patients with cancer-related conditions annually (approximately 70 patients per day). Their oncologic emergency medicine fellowship was then established in 2011 to provide additional training in the diagnosis, treatment, and multidisciplinary management of ED cancer patients [24]. This fellowship is 1 year (with an optional second year) dedicated to expanding the knowledge of those interested in pain and symptom management, palliative care, and oncologic emergencies. Each fellow works three 10-hour shifts per week in the ED under faculty supervision, completes inoffice research and administrative work 2 days per week, attends weekly didactic presentations for 2 hours, and attends faculty meetings and/or graduate medical education curricular lectures on a weekly basis. Second-year fellows pursue their selected clinical track and a research project. Fellow performance is assessed by the program director through biannual evaluations and faculty observation. Additionally, program feedback is obtained through biannual evaluation of the curriculum [23].

The James Cancer Hospital at The Ohio State University Wexner Medical Center opened its own oncologic ED in 2014. This 15-bed department sees approximately 13,000 patients with cancer each year and functions as an integrated component of their general ED. Their 1-year oncologic emergency medicine fellowship was established in 2017. Each fellow completes 7 months of training in the James ED and 1 month of education with each of the following services: oncologic, hematology, neuro-oncologic, and palliative medicine. Fellows also receive 1 month of protected time to conduct research and work on other scholarly projects. Additionally, each fellow can create their own niche and explore individual interests by pursuing an administration, research, or education path within the fellowship [23].

Continuing Medical Education

Beyond residency and fellowship, practicing physicians must continue to educate themselves in oncologic emergencies. This is important not only to provide quality patient care but also to properly educate learners on this topic at the bedside. Unfortunately, opportunities in continuing medical education specific to cancer-related conditions are limited. Most education focuses on febrile neutropenia, hypercalcemia in malignancy, superior vena cava syndrome, hyperviscosity syndrome, and tumor lysis syndrome, and there is little vetted information on more innovative topics such as immunotherapy toxicities. Nationwide, there are some educational opportunities within professional organizations' yearly conferences, such as the American College of Emergency Physicians (ACEP) Scientific Assembly or Society of Academic Emergency Medicine (SAEM) Annual Meeting. Additionally, there are two conferences dedicated specifically to oncologic emergency medicine: MD Anderson Cancer Center's Oncologic Emergency Medicine conference and Memorial Sloan Kettering Cancer Center's Emergencies in the Cancer Patient course.

References

- 1. Siegel RL, Miller KD, Jemal A. Cancer statistics, 2020. CA Cancer J Clin. 2020;70(1):7–30.
- Larkin J, Chiarion-Sileni V, Gonzalez R, Grob JJ, Rutkowski P, Lao CD, et al. Five-year survival with combined nivolumab and ipilimumab in advanced melanoma. N Engl J Med. 2019;381(16):1535–46.
- Sadik M, Ozlem K, Huseyin M, AliAyberk B, Ahmet S, Ozgur O. Attributes of cancer patients admitted to the emergency department in one year. World J Emerg Med. 2014;5(2):85–90.
- Thomas PA, Kern DE, Hughes MT, Chen BY. Curriculum development for medical education. 3rd ed. Baltimore: Johns Hopkins University Press; 2016.
- Accreditation Council for Graduate Medical Education, American Board of Emergency Medicine. The Emergency Medicine Milestones Project. July 2015. https://www.acgme.org/Portals/0/ PDFs/Milestones/EmergencyMedicineMilestones.pdf.
- Rivera DR, Gallicchio L, Brown J, Liu B, Kyriacou DN, Shelburne N. Trends in adult cancer-related emergency department utilization: an analysis of data from the nationwide emergency department sample. JAMA Oncol. 2017;3(10):e172450.
- Counselman FL, Babu K, Edens MA, Gorgas DL, Hobgood C, Marco CA, et al. The 2016 model of the clinical practice of emergency medicine. J Emerg Med. 2017;52(6):846–9.
- Common Program Requirements. Accreditation Council for Graduate Medical Education. (2020). https://www.acgme.org/ What-We-Do/Accreditation/Common-Program-Requirements. Accessed 21 Apr 2020.
- Accreditation Council for Graduate Medical Education. ACGME Program Requirements for Graduate Medical Education in Emergency Medicine. 2020. https://www.acgme.org/Specialties/ Program-Requirements-and-FAQs-and-Applications/pfcatid/7/ Emergency. Accessed 21 Apr 2020.
- Lamba S, Pound A, Rella J, Compton S. Emergency medicine resident education in palliative care: a needs assessment. J Palliat Med. 2012;15(5):516–20.
- Kraus CK, Greenberg MR, Ray DE, Dy SM. Palliative care education in emergency medicine residency training: a survey of program directors, associate program directors, and assistant program directors. J Pain Symptom Manag. 2016;51(5):898–906.
- Toohey SL, Wray A, Wiechmann W, Lin M, Boysen-Osborn M. Ten tips for engaging the millennial leaner and moving an emergency medicine residency curriculum into the 21st century. West J Emerg Med. 2016;17(3):337–43.
- Wolff M, Wagner MJ, Poznanski S, Schiller J, Santen S. Not another boring lecture: engaging learners with active learning techniques. J Emerg Med. 2015;48(1):85–93.
- Lew EK. Creating a contemporary clerkship curriculum: the flipped classroom model in emergency medicine. Int J Emerg Med. 2016;9(1):25.

- Neeley BC, Golden DW, Brower JV, Braunstein SE, Hirsch AE, Mattes MD. Student perspectives on oncology curricula at United States medical schools. J Cancer Educ. 2019;34(1):56–8.
- Mattes MD, Patel KR, Burt LM, Hirsch AE. A nationwide medical student assessment of oncology education. J Cancer Educ. 2016;31(4):679–86.
- Xu J, Murphy SL, Kockanek KD, Arias E. Mortality in the United States, 2018. NCHS Data Brief. 2020;355:1–8.
- Horowitz R, Gramling R, Quill T. Palliative care education in U.S. medical schools. Med Educ. 2014;48(1):59–65.
- Pavlidis N, Vermorken JB, Stahel R, Bernier K, Cervantes A, Pentheroudakis G, et al. Undergraduate training in oncology: an ESO continuing challenge for medical students. Surg Oncol. 2012;21(1):15–21.
- 20. Head BA, Schapmire TJ, Earnshaw L, Chenault J, Pfeifer M, Sawning S, et al. Improving medical graduates' training in pallia-

tive care: advancing education and practice. Adv Med Educ Pract. 2016;7:99–113.

- Halfdanarson T, Hogan W, Madsen B. Emergencies in hematology and oncology. Mayo Clin Proc. 2017;92(4):609–41.
- 22. Thistlethwaite JE, Davies D, Ekeocha S, Kidd JM, MacDougall C, Matthews P, et al. The effectiveness of case-based learning in health professional education: a BEME systematic review: BEME guide 23. Med Teach. 2012;34(6):e421–44.
- Purcell MG, Majzoub IE. The oncologic emergency medicine fellowship. Emerg Med Clin North Am. 2018;36(3):637–43.
- The University of Texas MD Anderson Cancer Center. Emergency Medicine. 2020. [Internet] https://www.mdanderson.org/research/ departments-labs-institutes/departments-divisions/emergencymedicine.html. Accessed 25 Apr 2020.



National and Institutional Research Efforts

Cielito Reyes-Gibby and Jenny L. Ren

Introduction

Advances in cancer diagnosis and treatment have led to increasing life expectancy of cancer patients. The number of cancer survivors in the United States has more than tripled over the past 30 years to over 16.9 million Americans [1]. As cancer evolves to a chronic disease, and with toxicities arising from cancer treatments including surgery, chemotherapy, radiation, and newer cancer therapies, both disease and symptom management of the cancer patient continue to increase in complexity. Many of these patients receive care in emergency departments as most cancer therapies are provided on an outpatient basis. Rapidly identifying the risk profiles of these patients, along with understanding the timing, sequence, duration, and treatment of disease processes and treatment effects, is the most important challenge faced by practitioners in oncologic emergency medicine.

The National Institutes of Health (NIH) invests about \$41.7 billion annually to support medical research in the USA [2]. The majority (80%) of this funding supports extramural research awarded through almost 50,000 competitive grants supporting more than 300,000 researchers from more than 2500 universities, medical schools, and other research institutions in every state. Table 69.1 shows the different NIH institutes and centers and their years of establishment [3]. Each fiscal year, these institutes or centers are required to prepare for the US President and US Congress its best professional judgment on the optimum funding needed to make the most rapid progress in their area of science, including funding research investigators, research training, and

C. Reyes-Gibby (⊠)

 Table 69.1
 Institutes and centers established at the National Institutes

 of Health (US Department of Health and Human Services)

		Year
Institutes and Center	Acronym	established
National Cancer Institute	NCI	1937
National Heart, Lung, and Blood Institute	NHLBI	1948
National Institute of Allergy and Infectious Diseases	NIAID	1955
National Institute of Dental and Craniofacial Research	NIDCR	1948
National Institute of Mental Health	NIMH	1949
National Institute of Neurological Disorders and Stroke	NINDS	1950
Eunice Kennedy Shriver national institute of child health and human development	NICHD	1962
National Institute of General Medical Sciences	NIGMS	1962
National eye Institute	NEI	1968
National Institute of Environmental Health Sciences	NIEHS	1966
National Institute on Alcohol Abuse and Alcoholism	NIAAA	1974
National Institute on Aging	NIA	1974
National Institute on Drug Abuse	NIDA	1974
National Institute of Arthritis and Musculoskeletal and Skin Diseases	NIAMS	1986
National Institute of Nursing Research	NINR	1986
National Institute on Deafness and Other Communication Disorders	NIDCD	1988
National Human Genome Research Institute	NHGRI	1989
National Institute of Biomedical Imaging and Bioengineering	NIBIB	2000
National Institute on Minority Health and Health Disparities	NIMHD	2000
Fogarty International Center	FIC	1968
National Center for Advancing Translational Sciences	NCATS	2011
National Center for Complementary and Integrative Health	NCCIH	1998

education of new investigators. In 1937, the National Cancer Institute became the first institute to be established. No institute is dedicated solely to emergency medicine. Advancing

Department of Emergency Medicine, Division of Internal Medicine, Department of Biostatistics, The University of Texas MD Anderson Cancer Center, Houston, TX, USA e-mail: creyes@mdanderson.org

J. L. Ren

Department of Emergency Medicine, Division of Internal Medicine, The University of Texas MD Anderson Cancer Center, Houston, TX, USA

K. H. Todd et al. (eds.), Oncologic Emergency Medicine, https://doi.org/10.1007/978-3-030-67123-5_69

the science of oncologic emergencies will require collaboration with scientists from different disciplines, building upon existing research infrastructure and investing on research training for ED practitioners. In this chapter, we review research focusing on oncologic emergencies and present the study limitations identified by the authors, present two ongoing research programs at a national and institutional level, and offer future directions.

Limitations of Published Oncologic Emergency Research

This review discusses common limitations of published oncologic emergency research. We searched PubMed using the following terms: {[Emergency (Title/Abstract) or EM (Title/Abstract) or EMS (Title/Abstract) or ED (Title/ Abstract)) and [Cancer (Title/Abstract) or Neoplasia (Title/ Abstract) or Oncolog* (Title/Abstract)] and limited to publications within the last 10 years. After review of 444 abstracts, we included 48 articles on cancer patients presenting to the emergency department (Table 69.2) [4-51]. Of note, knowledge gaps in emergency care of cancer patients have already been published in the 2016 "Cancer and Emergency Medicine: Setting the Research Agenda" that identified four main research opportunities and priorities to advance the understanding of oncologic emergency care: (1) collection of epidemiologic data, (2) care of patients with febrile neutropenia, (3) information on acute events such as dyspnea or pain, and (4) palliative care in the ED setting [4]. The topics of most, if not all, included articles in this review touch on one or more of these themes.

Table 69.2 Limitations of published studies on oncologic emergencies

Author-Year	Title	Study type, population	Findings	Limitations
Brown et al. 2016 [4]	The emergency care of patients with cancer: Setting the research agenda	Research agenda	Identified research opportunities and priorities to advance understanding of emergency care: (1) collect epidemiologic data, (2) care of patient with febrile neutropenia, (3) acute events like dyspnea and acute pain, and (4) palliative care in the ED	
Philip et al. 2018 [5]	The experiences of patients with advanced cancer and caregivers presenting to emergency departments: A qualitative study	Cross-sectional study, 19 patients with advanced CA and ED visit, 10 informal caregivers	ED presentations largely prompted by worsening symptoms or to expedite hospital admission – Many directed to ED by PCP. ED experience: Anxiety, uncertainty with communication, the general environment, symptom management delays. Long waits common. Patients felt relief at receiving care	Limited number of EDs, only two services, English-speaking only
Coyne et al. 2017 [6]	Application of the MASCC and CISNE risk-stratification scores to identify low-risk febrile neutropenic patients in the emergency department	Retrospective cohort study, 230 patients from 2 academic EDs	Presenting with chemo-induced febrile neutropenia, CISNE identified 23% as low risk, highly specific (98.3%) for low-risk cohort for all outcomes (inpatient LOS, upgrade in LOC, clinical deterioration, positive blood cultures, death). Median LOS shorter for low- vs. high-risk patients (3-day difference). MASCC score was much less specific (54.2%) in identifying a low-risk cohort	Misclassification bias (especially for patient symptoms), retrospective study based on chart review, missing data, both sites NIH CA centers, thus increased disease relative to general ED population
Adler et al. 2019 [7]	Validation of the emergency severity index (version 4) for the triage of adult emergency department patients with active Cancer	Analysis of observational cohort, 1008 patients from CONCERN (18 EDs)	ESI scores among ED patients with active CA indicate higher acuity than general ED population. ESI associated with patient disposition/ resource use. No significant association between ESI and non-ED based outcomes (hospital LOS, 30-day mortality)	Academic, urban setting. 21% of total ineligible, excluded as too ill or unable to consent, fewer high acuity patients. Resource use not well defined in ESI handbook, underestimated. English-speaking only

		Study type,		
Author-Year	Title	population	Findings	Limitations
Caterino et al. 2019 [8]	Analysis of diagnoses, symptoms, medications, and admissions among patients with cancer presenting to emergency departments	Observational prospective cohort study, 1075 adults with active CA from CONCERN (18 EDs)	Mean age 62 ($47\% > 64$), 51.8% female, 73.9% with CA treatment in preceding 30 days, 62.7% had advanced or metastatic CA. 5 most common ED diagnoses were symptom-related. 7.6% placed in observation, 57.2% admitted. 25% admissions had LOS of 2 days or less. Pain during ED visit present in 62.1%, mean pain score 6.4/10. Pain in 72.2% during prior week. ED opioids given to 59.1% with moderate/severe pain. Outpatient opioids for 47.4% patients (with pre-ED pain), including 57% with quite a bit or very much pain. ED nausea present in 31.3% (47.6 received antiemetics). Antibiotics given to 26.5% (73.3% admitted vs. 54.1% not receiving antibiotics)	Large, urban, academic centers, more CA centers. Community ED differs in patient characteristics, resources, outcomes. May have underestimated severity within participating academic institutions. Didn't enroll all presenting CA patients. Some too ill/couldn't participate. Many non-English- speaking CA patients with limited English report inferior treatment outcomes. Underestimated illness severity, 2 possible effects: (1) symptom severity/frequency, admission rate, hospital LOS, other variables related to severity may represent lower severity, and (2) ceiling effect as more challenging to improve outcomes in healthier group. Couldn't identify use of healthcare services in other hospitals, thus underestimated ED revisit/readmission rates
Mueller et al. 2016 [9]	Frequent emergency department utilizers among children with cancer	Retrospective cohort, PHIS database	Frequent utilizers account for 58% of ED visits. They differed from infrequent users by CA type: 39.3% of frequent users had ALL and 16% had CNS tumors (vs. 21.9%, 24.8%, respectively). Frequent use associated with age 5–9 or 1–4 or age < 1 vs. 15–19. Also Hispanic vs. white, non-Hispanics, urban residence. Few children received no meds, lab, imaging during ED visit	Possible patient undercount as some are diagnosed/treated as outpatients. Only tertiary facilities, possible underestimation of prior care
Richards et al. 2011 [10]	Palliative care symptom assessment for patients with cancer in the emergency department: Validation of the screen for palliative and end-of-life care needs in the emergency department instrument	Prospective observational cohort study, 53 patients	53% subjects male, age 24–88. Most common CA diagnoses were breast, colon, and lung. The SPEED instrument demonstrates reliability and validity	Patient population in tertiary center ED may not generalize to other EDs. English-speaking patients in tertiary center
Ahn et al. 2018 [11]	Comparison of the MASCC and CISNE scores for identifying low-risk neutropenic fever patients: Analysis of data from three emergency departments of centers in three continents	Retrospective, 571 patients from 3 tertiary CA ED centers in USA, UK, S. Korea	MASCC: 89.1% classified as low risk, CISNE: 10.5%. MASCC had more discriminatory power in detecting low risk. Risk scores should be used in conjunction with clinical judgment to identify patients suitable for outpatient management	Misclassification bias due to retrospective nature. Tertiary centers are less generalizable

(continued)

Author-Year	Title	Study type, population	Findings	Limitations
Delgado-Guay et al. 2016 [12]	Characteristics and outcomes of patients with advanced cancer evaluated by a palliative care team at an emergency center. A retrospective study	Retrospective case control, 400 patients	Median age 56 years old, 56% female, 61% white, no demographic differences between ED and inpatients. Median time from admission to palliative care 12 h for ED patients vs. 24 h for inpatients. For ED vs. inpatients, symptoms included uncontrolled pain (83%/45%), N/V/constipation (41%/19%), dyspnea (29%/19%). At follow-up, improvement in pain, sleep, Well-being, fatigue, anxiety, appetite, dyspnea, nausea, depression, drowsiness. After PC consults, discharge/admission destinations for ED patients were home (33%), home hospice (7%), inpatient hospice (4%), regional hospital floor (33%), and PC unit (23%). Median duration of hospitalization 92 h for hospitalized ED patients and 125 h for inpatients. Referral to PC from ED led to earlier delivery of PC with earlier control of symptoms. ED patients receiving PC consults had shorter hospitalizations than those receiving PC referral as inpatients	Retrospective – Impossible to estimate causality between PC service involvement in ED patients and outcomes. Highly specialized CA center, unique patients. Difficult to describe all interventions provided by PC to ED patients vs. inpatients. Need multicenter prospective studies at national/international levels
Banala et al. 2017 [13]	Discharge or admit? Emergency department management of incidental pulmonary embolism in patients with cancer: a retrospective study	Retrospective cohort, 193 patients	Selected ED CA patients with incidental PE can be treated safely with low-molecular-weight heparin and discharged. Of 193 patients, 70% discharged, 30% admitted. 30-day survival rate: 99% for discharged, 76% for admitted. 98% received ED anticoagulant (90% LMWH). Incident saddle PEs had higher 30-day mortality. Age, comorbidity, race, CA stage, tachycardia, hypoxemia, and incidental PE location associated with hospital admission	Single center, limited numbers. Retrospective. No randomization of treatment. Health outcomes related to PE not well distinguished from those related to advanced CA. Need larger RCT
Batalini et al. 2017 [14]	Cancer complaints: The profile of patients from the emergency department of a Brazilian oncologic teaching hospital	Cross-sectional, 277 ED visits	Pain was most common complaint (40% visits) > constitutional symptoms (17%) > GI complaints (11%). Abdominal pain most noted pain type (18.4%) with highest rate of recurrence > back pain. Cervical CA (14.8%), breast (11.6%), lung (7.6%). Majority of patients visited ED less than once a month	Single site. No staging data. Patients may visit ED for more than 1 complaint and symptoms overlap. No longitudinal follow-up. Importance of research to finding new therapies
Baugh et al. 2016 [15]	Emergency department management of patients with febrile neutropenia: Guideline concordant or overly aggressive?	Retrospective cohort, 173 ED visits to Dana-Farber Cancer Institute	25% of visits low risk, 75% high risk. Management overall was guideline concordant in 70%, discordant in 98% low-risk patients vs. 7% high-risk patients. Of 52 guideline- discordant cases, 83% involved low-risk cases with more aggressive treatment than recommended	Single-center study. Retrospective chart review. Model may lack important unmeasured covariates. Nonadherence could be due to awareness gap or factors not captured in chart review. Couldn't tell if management plan driven by ED or oncologist. Did not power the study to detect differences in clinical outcomes between risk groups

Author-Year	Title	Study type, population	Findings	Limitations
Brown et al. 2017 [16]	An exploration of medical emergency team (MET) intervention at the end of life for people with advanced cancer	Case control, 100 patients	Cohort without MET intervention had better quality of death score vs. MET patients. Within MET cohort, if MET influenced EOL decision- making ($n = 19$), had a significantly higher quality of death score vs. MET patients where MET did not influence care	Retrospective chart review. Misinterpretation of quality of death indicators. Single CA center. No record of EOL discussions, family members'/friends' perceptions of quality of death. Need prospective observational study to enhance understanding of MET in EOL CA care
Elsayem et al. 2017 [17]	Advance directives, hospitalization, and survival among advanced cancer patients with delirium presenting to the emergency department: a prospective study	Prospective cross-sectional observational study, 243 randomly selected CA patients from MD Anderson ED	Group A = delirium diagnosed by both CAM and MDAS; group B = MDAS only; group C = neither. Hospitalization rates for A 82%, B 77%, C 49%. ICU rates: 18%, 14%, 2%. Advance directives: 52%, 27%, 43%. Median overall survival: 1.23 month, 4.7 months, 10.45 months. Overall survival did not differ between groups A and B, but group C survival exceeded other groups. Delirium assessed by either CAM or MDAS was associated with worse survival and more hospitalization in patients with advanced CA. Many advanced ED CA patients with delirium lack advance directives	Single-center study. Number of patients small. Did not adjust for comorbidities or illness severity. Did not follow patients longitudinally to evaluate whether further interventions were done during hospitalization or if patients discharged to hospice
Elsayem et al. 2016 [18]	Delirium frequency among advanced cancer patients presenting to an emergency department: A prospective, randomized, observational study	RCT, 243 English- speaking patients with advanced CA from MD Anderson ED	Median age 62. 9% had CAM+ delirium, median MDAS score 14. Patients with delirium had poorer performance status than patients without (but 2 groups did not differ in other characteristics). 10% of age > 64 patients had CAM+ delirium vs. 8% age < 65. Mild delirium in 82%, moderate in 18%. Physicians identified delirium in 59% of CAM+ patients	Single center, small sample. Unblinded ED physicians to potential of delirium in their patients, thus bias favoring more frequent recognition. Selection bias: Excluded unstable patients, refusal, dementia. Could have higher delirium prevalence than those consenting, thus underestimates prevalence of delirium. Did not enroll patients at night
Elsayem et al. 2016 [19]	Presenting symptoms in the emergency department as predictors of intensive care unit admissions and hospital mortality in a comprehensive cancer center	Retrospective cohort, 9246 patients, 16,038 visits to MD Anderson ED	Main presenting symptoms: Pain, fever, dyspnea. 58% admitted to hospital at least once (range 1–13), 13% admitted to ICU at least once, 11% died during hospitalization. Independent predictors of hospital death: Presenting symptoms of respiratory distress or altered mental status, lung CA, leukemia, lymphoma, nonwhite race. Patients who died had longer LOS than those discharged alive	Retrospective, single center. No data on whether patients had advanced progressive or limited disease or whether they were treated for cure or palliation. No tumor staging. Used only main presenting symptoms, many patients had multiple or secondary symptoms. Systematic symptom assessment will allow better understanding of multiple symptoms and their impact on QOI
Kyeremanteng et al. 2019 [20]	Outcomes and cost of patients with terminal cancer admitted to acute care in the final 2 weeks of life: a retrospective chart review	Retrospective cohort, 130 patients who visited ED within 2 weeks of death, the Ottawa Hospital	85% had metastatic disease. 71% of admitted patients did not have advanced care directives. 18% receiving palliative care. Patients hospitalized for 7 days on average. Hospitalization costs 2.5 times estimated hospice cost	Retrospective, single-center study, small sample size. Chart review – Not all data recorded. Cost calculations based on crude estimates

(continued)

Author-Year	Title	Study type, population	Findings	Limitations
El Majzoub et al. 2019 [21]	Adverse effects of immune checkpoint therapy in cancer patients visiting the emergency department of a comprehensive cancer center	Retrospective review, 257 ED patients receiving immune checkpoint therapy from MD Anderson ED	25% of visits related to immune- related adverse effects. Diarrhea was most common reason for visit. Proportions of visits associated with diarrhea, hypophysitis, thyroiditis, pancreatitis, or hepatitis varied significantly by immune checkpoint therapy agent. Colitis associated with better prognosis; pneumonitis with worse survival	Single institution, limited size. Physicians performing data abstraction not blinded to outcomes or previous abstraction results when performing data confirmation. EMR may be erroneous. Underestimated true survival (used last confirmed contact when death unverified). Immunotherapy always changing (available agents, approved indications, patient characteristics)
Mueller et al. 2019 [22]	Identifying patient- centered outcomes for children with cancer and their caregivers when they seek care in the emergency department	Cohort, 26 caregivers or children with CA from IU Riley Hospital for children Heme-Onc clinic	More important outcomes included system-level issues (e.g., cleanliness, timeliness) and oncologic-provider or ED-provider level issues (ability to access port-a-caths, quality of communication). Also identified outcomes that were within control of patient/caregiver, such as improving their sense of preparedness	Need future research on development and validation of patient-centered outcomes tool. Single institution – Data may not be representative. Small sample size
Peyrony et al. 2020 [23]	Antibiotic prescribing and outcomes in cancer patients with febrile neutropenia in the emergency department	Cohort, 249 patients	Median age 60, 67.9% hematological malignancy, 10.4% admitted to ICU, 9.8% died during hospital stay. 32.4% of low-risk patients presented at least 1 complication, including 11 deaths. Time to antibiotic initiation in ED not associated with outcome after adjusting for performance status and shock index. Inadequate ED antibiotic regimen associated with higher ICU admission and death during hospital stay (OR = 3.5). An inadequate ED antibiotic regimen in patients with FN was significantly associated with higher ICU admission or death during hospital stay	Retrospective, interpretation bias during chart abstraction. Antibiotic appropriateness may vary between centers. Possible missed confounders. Single-center – External validation needed
Reyes-Gibby et al. 2017 [24]	Cohort study of oncologic emergencies in patients with head and neck cancer	Prospective cohort, 298 patients	History of HTN, normal or underweight BMI, and probably depression predicted increased risk for ED presentation. BMI and severe pain associated with higher frequency of ED presentations	Small sample size. HPV status missing. Limited to patients with HNSCC from 1 tertiary care CA center. Possibly did not capture all ED visits (e.g., other hospital ED). Lack of data on etiology of pain
Reyes-Gibby et al. 2016 [25]	Risk for opioid misuse among emergency department cancer patients	Cross-sectional, 209 patients	On basis of SOAPP-R (screener and opioid assessment for patients with pain-revised) cutoff of 18, 34% patients had high risk of misuse. 15% and 4% of all patients reported past or current use of illicit substances, respectively. Total number of annual opioid prescriptions differed between high- and low-risk groups. Depression, poor coping, and illicit substance use were associated with high risk of opioid misuse	Single center. Patients already receiving opioids – May have affected survey responses. Patient self-report. Number of opioid prescriptions is imperfect representation of actual dosage/pill count. Future studies of patients with CA-related pain should include multiple measures of risk (risky medication-related behaviors, structured interview). Potentially inaccurate alcohol and smoking status. Nonrespondents had higher pain scores

		Study type.		
Author-Year	Title	population	Findings	Limitations
Tanriverdi et al. 2014 [26]	Single center experience on causes of cancer patients visiting the emergency department in Southwest Turkey	Retrospective cohort, 102 patients, 304 ED visits	65% patients male, 52% over age 65. 30% lung CA, 32% with dyspnea, 53% with metastasis, 30% with multiple metastatic lung lesions, 68% had poor ECOG performance status	Small sample size, single center
Yang et al. 2016 [27]	Cardiac troponin is a predictor of septic shock mortality in cancer patients in an emergency department: a retrospective cohort study	Retrospective cohort, 375 ED CA patients with septic shock	Creatine kinase myocardial band fraction and troponin-I significantly higher in patients who died in ≤7 days and ≤28 days than in those who did not. The SOPED (septic oncologic patients in emergency department) scoring system, which incorporated troponin-I, was more prognostically accurate than were other scores for 7-day mortality	Single center, small sample size, retrospective. Results should be verified in large well-designed multicenter clinical study. Advances in management of sepsis occurred during study time. Prognosis of CA patients with septic shock probably not generalizable to non-CA
Lamb et al. 2019 [28]	Hodgkin lymphoma detection and survival: Findings from the Haematological malignancy research network	Cohort, 971 patients	Median diagnosis at age 41.5, 55.2% male, 31.2% stage IV. 43% had moderate to high or high-risk prognostic score. 18.7% admitted via ED prior to diagnosis. Relationship between age and ED admission was U-shaped: More likely in patients <25 and > = 70. Compared to patients admitted via other routes, those via ED had more advanced disease and poorer survival. After adjusting for clinically important prognostic factors, no difference in survival remained	Lack of primary data, could not investigate patterns or referrals from primary care, Hodgkin lymphoma is rare
Mayer et al. 2011 [29]	Why do patients with cancer visit emergency departments? Results of a 2008 population study in North Carolina	Observational, 2008 NC DETECT ED visit data, 37,760 ED visits, 27,644 CA patients	77.2% of patients had 1 ED visit in 2008. Mean age 64, slightly more men. Among visits, Medicare 52.4%, Medicaid 12.1%. >1/2 visits weekends or evenings, 44.9% during office hours. Top 3 complaints: Pain, dyspnea, GI issues. Lung, breast, prostate, and colorectal CAs had largest proportions. 63.2% visits resulted in hospital admission. Patients with lung CA were more likely to be admitted than other CA types	All data collected for other purposes. No associations between county of residence, site of usual healthcare, and location of ED (NC DETECT didn't allow identifying information). No data on whether patients were under clinical care or where. No race/ethnicity data. Some visits that were CA-related may have been missed, some visits that were not CA-related may have been included. Not possible to identify visits by patient to multiple EDs. Limitation to categorization of presenting symptoms (not exhaustive/completely inclusive). Not possible to tell if some patients visited local EDs because routine care was at a CA center far from home. If healthcare services had been available, could those patients have avoided an ED visit? Need more info about these patients before and after their visits

(continued)

Author-Year	Title	Study type, population	Findings	Limitations
Mueller et al. 2016 [30]	Characteristics of children with cancer discharged or admitted from the emergency department	Retrospective, 26,770 ED visits, 17,943 children with CA from 39 children's hospitals	Half of children with CA visited ED within 1 year after their first CA hospitalization in pediatric health information system (PHIS). 56% of ED visits resulted in admission. Fever and neutropenia were most common reasons for visits (34.6%). Risk factors: "Other" race/ethnicity, history of transplant, and ED visits reasons including neutropenia, bloodstream infection, pancytopenia, dehydration, or pneumonia	Some patients diagnosed and treated as outpatients. PHIS does not include data on disease status, treatment, or time since last treatment for patient when they visit ED. discharge ICD-9 codes limits understanding of chief complaints to ED, thus lacks key data for anticipatory guidance. Extrapolated reasons for ED visit from ICD codes rather than the initial complaint. Only single diagnosis evaluated per patient encounter, but children with CA may present to ED with multiple problems
Mueller et al. 2020 [31]	Variation in hospital admission from the emergency department for children with cancer: a pediatric health information system study	Retrospective cohort, 60,054 ED visits for children with CA, using PHIS data	62.5% admitted, primary diagnosis – Fever. Largest variability in admission rates. Less variability among hospital admission rates for neutropenia and febrile neutropenia. Admission rates by day of the week did not demonstrate significant variability and no differences seen between weekend and weekday	ICD codes may not accurately identify all CA patients, leading to over/underestimate of ED cases. PHIS data set did not reliably detect ED chief complaint – Extrapolated from ICD9 codes. Variability in coding by hospital, PHIS did not include all hospitals that treat children with CA. No access to important clinical data. Did not account for holidays
Patel et al. 2017 [32]	Evaluation of emergency department management of opioid-tolerant cancer patients with acute pain	Retrospective cohort study, 216 patients Opioid-tolerant CA patients who received opioids in ED over 2 years	61.1% of ED patients received adequate initial PRN dose of opioids. Of patients taking <200 OMEs per day at home, 77.4% received an adequate initial dose. 3.2% of patients taking >400 OMEs per day at home received an adequate dose. Patients with ambulatory 24-hour OME greater than 400 had 99% lower odds of receiving an adequate initial dose of PRN opioid in the ED compared to patients with ambulatory 24-hour OME less than 100	Retrospective study. Assumed patients were taking meds as prescribed. May have been skewed to higher ambulatory use. Not all ED visits studied were primarily for pain-related complaints. Didn't assess whether CA was active/ inactive during study period. Not enough patients at extremes of age pain scores not routinely documented
Sadik et al. 2014 [33]	Attributes of cancer patients admitted to the emergency department in one year	Retrospective review, 408 ED CA patients	58.8% male, median age 57.9, 65.3% had metastatic disease. Hospitalization rate 59.6%. Most common symptoms were dyspnea, pain, fever, and N/V. Most common CA sites were lung, GI, and breast. Initial evaluation found progressive disease, chemotherapy effects, infections, radiotherapy effects, extravasation, anemia, and unknown. During follow-up, 46.8% patients died after admission to ED. 1-year overall survival of all patients was 7 3 months	Retrospective. Survival data collected from diverse sources, including telephone surveys
923

		Study type,		
Author-Year	Title	population	Findings	Limitations
Seow et al. 2016 [34]	Does increasing home care nursing reduce emergency department visits at the end of life? A population-based cohort study of cancer decedents	Retrospective cohort, 54,576 EOL cancer decedents in Ontario, Canada	Of decedents using home care nursing services in last 6 months before death, 85% had ED visit, and 68% received EOL home care nursing. Patients receiving EOL nursing vs. standard nursing had reduced ED rate. In last month of life, receiving EOL nursing and standard nursing rate of >5 hours/week was associated with fewer ED visit rates of 41%. Temporal association between receiving end-of-life nursing in a given week during the last 6 months of life and of more standard nursing in the last month of life was associated with a reduced ED rate in the subsequent week	Retrospective – No causality. Administrative databases limited – No data on caregiver support, patient clinical/psychosocial, physician care, symptom severity, or privately obtained home care services. Does not analyze reasons for ED visits. Canadian system
Scholer et al. 2017 [35]	Improving cancer patient emergency room utilization: A New Jersey state assessment	Retrospective cohort, 37,080 ED CA visits in New Jersey	Most frequent diagnosis: Lung CA (30%), most common complaint: Pain. Patients who visited ED predicted by race, age 65–75, number of diagnosis, insurance payer, CA type. Comorbidities increase mortality, being transferred to SNF/ ICF, using home healthcare services. Readmission is affected by race, income, and type of CA	No causality. Secondary data collected for other purposes. No hospital/ED facility ID information or revisits. Need ability to link healthcare data from SEER, Medicare, Medicaid, to allow a more comprehensive view on participating events, sequelae after ED visit, and effect of multimodal treatment
Qdaisat et al. 2020 [36]	Evaluation of cancer patients with suspected pulmonary embolism: Performance of the American College of Physicians Guideline	Retrospective, 380 patients	56% received CT pulmonary angiogram (CTPA) per ACP guideline, 21% received CTPA not per guideline. Pulmonary embolisms (PEs) were in 6% low-risk, 10% intermediate-risk, and 25% high-risk patients. ACP guideline had NPV of 99% and sensitivity of 97% in predicting PE	Retrospective. Inclusion of only patients who underwent CTPA
Grudzen et al. 2016 [37]	Emergency department- initiated palliative care in advanced cancer: a randomized clinical trial	RCT, 136 patients	QOL higher in palliative care group. Median estimates of survival were longer in PC group, but not significantly. No statistically significant differences in depression, admission to ICU, or discharge to hospice	Variable length of survival in cohort. Need future trial that limits enrollment to patients at similar CA stage. Unable to ascertain whether ICU admit/hospice referral was a reflection of patient's goals of care or if reflection of what team offered the patient. Missing data – Many patients did not survive for follow-up

(continued)

		Study type,		
Author-Year	Title	population	Findings	Limitations
Vandyk et al. 2012 [38]	Emergency department visits for symptoms experienced by oncologic patients: a systematic review	Systematic review, 18 studies cited in Medline, Embase, PsycINFO, and CINAHL	Of 28 symptoms reported, most common were febrile neutropenia, infection, pain, fever, and dyspnea. Definitions provided for individual symptoms were inconsistent. ED visits resulted in hospital admissions 58% (median) of the time in multi-symptom studies and 100% of the time in targeted symptoms studies. 13% (median) of ED visits captured in multi-symptom studies (range 1% to 56%) and 20% (median) of visits in targeted symptoms studies (range 4% to 67%) resulted in death	Few symptoms were defined adequately to compare data across studies (gap in CA symptom reporting). Symptom definitions inconsistent or not provided. Combined prevalence of any symptom could over–/under- represent actual ED use. Need better use of standardized definitions to improve reliability/ validity of reporting/facilitate synthesis across studies. Need research exploring the impact of risk of exposure to communicable disease on health of oncologic patients visiting EDs. Sample sizes varied significantly. Sometimes repeat visits by single patients were excluded possible that patients who needed ED care most were under-represented. Mortality inconsistently reported (rates collected at different intervals). In most articles, unclear whether authors identified confounding factors or if analyses took them into account
Baugh et al. 2019 [39]	Near-universal hospitalization of US emergency department patients with cancer and febrile neutropenia	Observational, 348,868 ED visits	94% of ED visits resulted in hospitalization. Private, self-pay, and other insurance were less likely to be hospitalized than those with public insurance. Hospitalization was least likely at non-metropolitan hospitals and metropolitan non-teaching hospitals. 26% of variability in hospitalization rate was attributable to which hospital the patient visited	Case ascertainment depended on diagnostic codes, which may be subject to error. Codes used to specify neutropenia did not specify a definition. Lack of a single discharge code describing FN may have also led to an underestimate of the true number of cases. NEDS does not track revisits by the same patient
Huang et al. 2020 [40]	Review article: End-of- life care for older people in the emergency department: A scoping review	Review, 14 articles	Definitions pertaining to EOL care in the ED vary. Older people presenting to ED at EOL were mostly female, triaged in urgent or semi-urgent category, presented with diagnoses of advanced CA, cardiac and pulmonary disease, and dementia with symptoms including pain and breathlessness. Multiple tools pertaining to EOL exist and range from predicting mortality and assessing functional status, comorbidities, symptom distress, palliative care needs, quality of life, and caregiver's stress. Outcomes for older people enrolled in specific EOL intervention programs included lower admission rates, shorter ED length of stay, increased palliative care referral and consultations, and decreased Medicare costs	Limited evidence exists regarding the definition, clinical profile, care delivery and outcomes for older people requiring EOL care in the ED

Author-Year	Title	Study type, population	Findings	Limitations
Lee et al. 2015 [41]	Emergency visits among end-of-life cancer patients in Taiwan: a nationwide population- based study	Retrospective cohort, 32,772 ED CA patients in Taiwan	81.5% ED visits in mortality group; higher ED utilization significantly monthly to EOL. Most frequent CA types were digestive and peritoneum CA (35%), breast (17.7%), and H&N (13.3%). Older patients, males, metastatic disease, and respiratory or digestive CA were more likely to use ED services at EOL. Use of an ED service in nearest community hospital to replace centers for dying CA patients would be more acceptable in emergency situation	If patient suffered prolonged hospitalization in last 6 months of life, may underestimate ED utilization. Loss of detail in claims data mining. Possible confounders not controlled, including socioeconomic status and family support
Mills et al. 2019 [42]	Factors affecting use of unscheduled care for people with advanced cancer: a retrospective cohort study in Scotland	Retrospective cohort, 2443 CA deaths in Tayside, Scotland	Of those dying from CA, 77.9% attended unscheduled care in the year before death. Among these, 10.7% A&E only and 33.1% both (along with GP). 19.7% attendances in last week, 36.7% last 4 weeks, 60.3% last 12 weeks of life. Age, sex, deprivation, and CA type not associated with unscheduled care attendance. Rural areas less likely to attend unscheduled care. Pain was most frequent coded clinical reason for presenting. Of those dying from CA, 21% were frequent users (> = 5 attendances/year) and account for over half of unscheduled care	Variable clinical coding of reason for attendance. Non-specific/ missing coding
Panattoni et al. 2018 [43]	Characterizing potentially preventable cancer- and chronic disease-related emergency department use in the year after treatment initiation: a regional study	Cohort, 5853 eligible patients, 2400 visits within 1 year of treatment	27% had at least 1 ED visit. 49.8% of ED visits had a potentially preventable diagnosis. The prevalence of potentially preventable ED visits was generally high, but varied depending on the diagnosis code fields and the group of codes considered	Claims records are imperfect proxies in CA. Validation studies needed to determine best uses and methodological approach to measuring potentially preventable ED use. Commercially insured population – can't extrapolate to Medicare, Medicaid, or other regional groups. Excluded ED visits with a direct transfer to inpatient hospitalization, likely undercounts all potentially preventable adverse outcomes. Other issues beyond symptoms may influence ED use – Fear, cultural background, insufficient language/ communication skills, delays in seeking help, nonadherence, and lack of social support

(continued)

Author Vear	Title	Study type,	Findings	Limitations
Rivera et al. 2017 [44]	Trends in adult cancer- related emergency department utilization: An analysis of data from the nationwide emergency department sample	Observational, 7 survey cycles from Nationwide emergency department sample (NEDS), 29.5 million CA patients, 696 million adult ED visits	Of all adult ED visits, 4.2% were by patients with CA. Most common were breast, prostate, and lung CA. Most common primary reasons for visit were pneumonia, nonspecific chest pain, and urinary tract infection. Adult CA-related ED visits resulted in inpatient admissions more than non-CA-related visits. Septicemia and intestinal obstruction were associated with highest odds of inpatient admits	Inherent limitations of data set. If a patient with CA visited ED and CA code not documented, then patient was misclassified into nonCA visit group. No separate CA staging/ treatment codes in NEDS data set other than maintenance chemo or radiotherapy, thus not possible to distinguish type of treatment or whether patients were receiving active treatment, already completed, or advanced disease. Lack of detailed staging and treatment data – Unable to examine reasons for CA-related visits at points along CA continuum. Primary reason for visit based on codes – Not enough detail. Need more detailed data on reason for visits to ED and those related to patient's CA (date of diagnosis, CA type/stage, previous treatment, and clinical outcomes)
Tang et al. 2017 [45]	An analysis of emergency department visits and the survival rate for colorectal cancer (CRC) patients: a nationwide population-based study	Retrospective cohort, 6532 ED visits by 3347 colorectal patients in Taiwan	Top three most common reasons ED visits: Ill-defined conditions, abdominal pain, and intestinal obstruction. Overall survival rates of colorectal patients in the ED visit group at 3, 5, and 10 years were 0.65, 0.56, and 0.47, respectively, without significant differences from rates among colorectal CA patients who did not visit the ED	Did not differentiate end-stage patients from general patients with CRC. Unable to detect actual reason for ED visit. ICD system designed for general purposes, not fit to describe CA-associated problems
Liao et al. 2017 [46]	Effects of multidisciplinary team on emergency care for colorectal cancer patients: A nationwide-matched cohort study	Observational study, 45,418 patients with newly CRC diagnosis, Taiwan National Health Insurance Research Database	Odds ratio (OR) by probability of emergency care used for participation in multidisciplinary care groups (MDT) within a year of CA diagnosis was less than that for nonparticipation. Significant benefits of MDT in CRC care	Secondary database. Patient ED care by urgency level not addressed
Yap et al. 2018 [47]	Patterns of care and emergency presentations for people with non-small cell lung cancer in New South Wales, Australia: A population-based study	Retrospective cohort, 647 NSCLC cases in Australia	58.6% male, median age 73. ED presenters (34.5% of cases) were more likely to have a high Charlson comorbidity index score, be an ex-smoker who had quit in the past 15 years, and be diagnosed with distant metastases. Almost all patients had visited their general practitioner ≥3 times in the 6 months prior to diagnosis. Nearly one-third (29.5%) of patients did not receive any anti-CA treatment; however, there were no differences between emergency and non-emergency presenters in the likelihood of receiving treatment. Those less likely to be treated were older, had no private health insurance, and had unknown stage disease recorded	45 and older sample, not whole NSW population. Could not directly infer that patients who visited ED in month/month prior to diagnosis were detected as a result of ED presentation. No access to more detailed TNM staging or ECOG performance status. 19% unknown stage

		Study type,		
Author-Year	Title	population	Findings	Limitations
Hryniewicki et al. 2018 [48]	Management of immune checkpoint inhibitor toxicities: a review and clinical guideline for emergency physicians	Review, 50 articles via PubMed	Inhibition of immune checkpoints may lead to loss of peripheral tolerance and subsequent unleashing of immune system on nontumor cells, leading to unintended tissue damage and multisystem organ dysfunction. Most commonly affected organ systems are dermatologic, GI, endocrine, and pulmonary. Treatment can range drastically depending on severity of irAE (immune-related adverse events): Supportive care, high-dose steroids, additional immune modulators (infliximab, IVIG)	Further studies needed to analyze outcomes of different treatment strategies in patients presenting with potential immune-related adverse events
Knight et al. 2017 [49]	Acute oncologic care: A narrative review of the acute management of neutropenic sepsis and immune-related toxicities of checkpoint inhibitors	Review	Outpatient management of low-risk febrile neutropenia patients identified by the MASCC score is a safe and effective strategy. Immunotherapy with "checkpoint inhibitors" has significantly improved outcomes for patients with metastatic melanoma, and evidence of benefit in a wide range of malignancies is developing. Timing of the onset of the adverse events is dependent on the organ system affected and unlike anti- neoplastic therapy can be delayed significantly after initiation or completion of therapy	Further research into optimal management, strategies, and pathways of acutely ill CA patients is required
Bowers et al. 2017 [50]	Ketamine as an adjunct to opioids for acute pain in the emergency department: a randomized controlled trial	RCT, 116 patients	Patients receiving ketamine reported lower pain scores over 120 minutes than patients receiving placebo. Total opioid dose was lower in the ketamine group. Satisfaction did not differ between groups. Fewer patients in the ketamine group required additional opioid doses. More patients reported light-headedness and dizziness in the ketamine group	Cohort with chronic pain and long-term outpatient opioid use may react differently to the adjunctive ketamine usage than patients with new-onset acute pain. ED environment allows for less precise control of variables. Subjectivity of patient-reported values. Sometimes a time point was missed
Northfield et al. 2019 [51]	Taking care of our own: A narrative review of cancer care services-led models of care providing emergent care to patients with cancer	Review, 22 studies	Overarching outcomes associated with the most commonly described models of care (telephone advice services and/or unplanned care and assessment units) were improved coordination of care/continuity of care, prompt access to specialist care, reduced utilization of EDs, fewer hospital admissions, and reduced cost	Methodological weaknesses of included studies and variations in study designs, settings, and models of care. Future studies should explore patient and caregiver CA experiences

CA cancer, CAM Confusion Assessment Method, CISNE Clinical Index of Stable Febrile Neutropenia, ECOG Eastern Cooperative Oncologic Group, EOL end of life, ESI emergency severity index, HNSCC head and neck squamous cell carcinoma, HTN hypertension, LMWH lowmolecular-weight heparin, LOS length of stay, MASCC Multinational Association of Supportive Care in Cancer, MDAS Memorial Delirium Assessment Scale, NIH National Institutes of Health, OME oral morphine equivalent, RCT randomized control trial, SEER Surveillance, Epidemiology, and End Results Program The most commonly cited limitation of all articles was generalizability. Majority of studies were conducted in an academic, urban setting (community hospital ED cancer prevalence and care may differ) [5–8] at tertiary cancer centers [9–12] or a single center (which may have limited the number of patients in the study) [13–27]. Selection bias was identified for including only English-speaking patients [5, 7, 8, 10]. Furthermore, patients that were too ill or unable to consent were deemed ineligible, which may have underestimated the severity of patients presenting to the ED [7, 8, 18, 25]. As a result, many studies reported the lower bounds of severity and acuity and therefore may contribute to a ceiling effect (when it is more challenging to improve outcomes in a healthier subset) [8].

Most studies were retrospective [6, 9, 11–13, 15, 16, 19– 21, 23, 26–36], often relying on chart review, which may lead to misclassification bias when data were misinterpreted or missing [6, 11, 16, 20, 21, 23, 25, 28, 29, 36, 37]. Additionally, many of the studies reported a lack of longitudinal follow-up, complicated by different timing and measures of mortality rates [14, 17, 21, 29, 35]. Some studies claimed they could not track revisits by the same patient, possibly counting some patients more than once, or they were unable to identify visits by the same patient to different EDs [24, 29, 38, 39].

Another common limitation identified by the authors was the inaccuracy and/or inadequacy of diagnostic codes - there was no standardization of definitions of chief complaints, as well as limited categorization [19, 29-31, 38-45]. Therefore, the true number of cancer patients presenting to emergency centers may not be accurately represented [31, 39, 44]. It was also difficult to account for multiple symptoms, as health records and ICD-9 codes often only reported the main presenting symptom used to assess ED admissions [14, 19, 30]. The databases used for many of these retrospective studies were secondary data and thus limited in the types of information recorded [28, 29, 34, 35, 46]. Many articles reported lack of staging information [14, 19, 44, 47] and disease or treatment status [19, 30-32, 37, 44, 45, 47] (e.g., advanced, progressive, or limited disease, new diagnosis or continued care, active or inactive cancer). It was also unclear whether multiple models lacked unmeasured confounders or if the analyses were adjusted for their potential confounding effect [15, 23, 38, 41, 43]. This would include factors such as socioeconomic status, family support, cultural background, delays in seeking help, medical nonadherence, and lack of social support.

Other limitations focused on the evolving field of cancer treatment, which may have altered findings throughout the duration of the study. For example, two articles focusing on immunotherapy discussed the rapidly changing field and newly identified immune-related adverse events [21, 48]. Another two articles pointed out that sepsis management of

cancer patients in the ED also changed during the respective studies [27, 49]. Multiple papers also called for novel research on pain in patients with cancer presenting to the ED [14, 24, 25, 32, 50]. These authors pointed out that their population had a mix of chronic and acute pain complaints, and their results may have been confounded by possible opioid tolerance in some patients [25, 50]. In addition, pain scores were not routinely documented [32] and no data had been published on the etiologies of cancer pain [24]. Patient-reported pain scores and satisfaction levels also have an inherently subjective nature [25, 50]. Lastly, some publications identified the gap in knowledge on patients' goals of care and how that may influence their experience in the ED [37, 51].

Overall, there is a need for larger, prospective cohort studies of oncologic patients presenting to the ED that would allow for understanding readmissions/revisits and rehospitalizations, multicenter randomized control trials, and inclusion of diverse populations. There is also a need for more accurate, standardized recording of diagnoses of patients presenting to the ED, including a systematic assessment of those with multiple symptoms. Given the rapid advances in cancer treatment, developing risk profiles of cancer patients, along with understanding the timing, sequence, and duration of toxicities of cancer treatment, is needed.

Training Opportunities

Given the large amount of existing federal funding in support of cancer research, there are many opportunities for academic emergency physicians to partner with their colleagues in other specialties who study cancer-related emergencies. This is particularly true for academic departments of emergency medicine situated in one of the 41 National Cancer Institute-designated comprehensive cancer centers. These centers offer a number of institutional research training programs (e.g., National Research Service Awards – T32 grants) that can deliver formal research training to junior academic emergency physicians who wish to pursue traditional research careers.

The Comprehensive Oncologic Emergencies Research Network (CONCERN)

In March 2015, the National Cancer Institute and the Office of Emergency Care Research sponsored a 1-day workshop, titled "Cancer and Emergency Medicine: Setting the Research Agenda," in Bethesda, MD. The goal of the workshop was to identify research opportunities and determine research priorities for issues related to the emergency care of

Topic ^a	Areas of concern			
Emergency department (ED) use by cancer patients	Definition of a cancer patient	Definition of a cancer- related ED visit	Electronic medical records	Options for ED care
Febrile neutropenia	Definition of febrile neutropenia; objective response	Optimal treatment strategies	Effects of timing of antibiotic therapy	Biomarker and risk stratification
Acute pain	Optimal treatment strategies	Coordination of care with outside providers	Barriers, skills and attitudes of ED providers	Biomarkers and risk stratification
Palliative care	Use of life-sustaining therapies	Disposition pathways/ follow-up care	ED and community hospice partnership	Larger cohort studies

Table 69.3 Partial list of workshop recommendations from the March 25, 2015, workshop titled "Cancer and Emergency Medicine: Setting the Research Agenda," in Bethesda, MD, sponsored by the National Cancer Institute and the Office of Emergency Care Research

^aAcute dyspnea and spinal cord syndrome were also among the acute events discussed

the cancer. Twenty-six participants representing the fields of emergency medicine (8), oncologic (3), internal medicine (1), oncologic EDs (4), operations (1), and palliative care/ emergency medicine (3) were invited to the workshop. Representatives from the National Cancer Institute (4), the National Institute of Nursing Research (1), and the Office of Emergency Care Research (1) were also in attendance and participated in the discussion. The results of this one day workshop have been published [4]. Table 69.3 shows some of the research recommendations from the workshop (not an exhaustive list) (https://epi.grants.cancer.gov/events/ emergency-medicine/).

This 1-day workshop led to the formation of CONCERN. Currently, more than 20 medical centers are represented. Membership is open to new members interested in collaborative research in cancer and emergency medicine. There are no membership requirements [52].

CONCERN objectives include: (1) identify knowledge gaps in the emergency care of cancer patients and define the scope, (2) accelerate knowledge generation and translation through multicenter research collaborations across oncologic and emergency medicine, (3) expand knowledge around treatment of oncologic emergencies in the emergency medicine setting, (4) translate research findings into national guidelines and community emergency care settings, (5) develop collaborative educational efforts in the area of cancer and emergency medicine, and (6) provide mentoring and consultation to new and established investigators working in the area of cancer and emergency medicine.

The first project included a multi-institutional prospective data collection which has been completed. The results have been published [7, 8], with additional publications currently ongoing. Periodic meetings via teleconference allow members to pursue collaborative projects and potential funding mechanisms are discussed. To date, plans are underway for a workshop at the National Institutes of Health campus in 2020. The goal of the workshop is to provide a 5-year research progress update, set new/revised research priorities, and establish new research collaborations at the intersection of cancer and emergency medicine. Priority Setting Questions to be addressed within each session include:

- 1. What are the gaps in knowledge/science around prevention and management of cancer-related emergency department visits?
- 2. What are specific research questions to address these gaps?
- 3. What are the resources, collaborations, and support needed to move this science forward?

Program in Oncologic Emergency Medicine (POEM), Department of Emergency Medicine at MD Anderson Cancer Center

Established in 2010, MD Anderson's Department of Emergency Medicine (DEM) is the first academic department in a comprehensive cancer center dedicated to the development of oncologic emergency medicine as a distinct discipline. The department's vision is to be the premier resource for education, research, and clinical care in the emerging discipline of oncologic emergency medicine. DEM presents a unique and exciting opportunity for exploring areas of research to improve the care of cancer patients and to identify gaps in current diagnostic and treatment approaches. The emergency center provides emergent and urgent treatment for MD Anderson patients and for employees and visitors injured on the MD Anderson premises. Every day, the emergency center provides service to approximately 70 patients with cancer-related conditions and is a vital safety net for MD Anderson patients. The emergency center features 44 rooms, including those designated for isolation, cardiovascular emergencies, and gynecologic emergencies. It is staffed by Department of Emergency Medicine physicians, mid-level providers, registered nurses, patient-service coordinators, and patient-care assistants.

In 2015, the MD Anderson Program in Oncologic Emergency Medicine was established to improve patient outcomes by providing an interdisciplinary research hub that serves as a framework for developing emergency care research and implementing research-driven practice in caring for cancer patients who present in the emergency center. The Program integrates existing and newly developed research initiatives, with reduction in oncologic emergencies as its primary focus.

The Program's mission is to improve the care of patients with cancer and to reduce their need for emergency care during survivorship. Its vision is to be a locus for promoting evidence-based multidisciplinary and translational research in oncologic emergencies for a broad spectrum of clinicians and researchers across the institution and beyond. The Program is highly complementary to the disease-specific and treatment-specific centers at MD Anderson, none of which support specific research in oncologic emergencies.

Program Goals

POEM's program goals are to (1) facilitate the conduct of synergistic and collaborative projects focusing on oncologic emergencies, (2) develop and train physician scientists and graduate students in oncologic emergency research, (3) accelerate knowledge generation and synthesis by bringing together clinicians and scientists from diverse disciplines, and (4) establish synergy with existing centers and departments at MD Anderson.

To Facilitate Synergistic and Collaborative Projects Focused on Oncologic Emergencies

MD Anderson offers a breadth of resources to the Program, ranging from existing research and clinical databases, a large patient population, a substantial "core" support infrastructure, and multifaceted, experienced faculty. Publicly available databases can augment institutional data repositories.

Data Repositories Program researchers can capitalize on existing databases within the department and institutional data sources, including genomics, tumor registry, pharmacy, bioinformatics and biostatistics, and radiology and laboratory data, along with patient demographic, disease, and treatment data. The Program links these various databases together to create a powerful, comprehensive repository of personalized patient data (Fig. 69.1). Research protocols approved by MD Anderson's Institutional Review Board facilitate data collection and analysis. For example, prior to MD Anderson's adoption of Electronic Privacy Information Center (EPIC) for patient data across the institution, a locally developed database (the Zero G/Whiteboard) was utilized for collection of ED patient data. It included, for example, demographic information, body mass index and weight loss, type of cancer, primary and secondary presenting symptoms (chief complaints), number and frequency of emergency visits, symptom severity, and final disposition (e.g., admitted to hospital or discharged home). These data were linked to other institutional databases (see Fig. 69.1).

To Develop and Train Physician Scientists and Graduate Students in Oncologic Emergency Medicine Research

PubLab and Manuscript Advancement and Development Group (MAD): POEM's objectives were to build the literature in oncologic emergency medicine to improve patient care, and to build upon the expertise of DEM clinical faculty as thought leaders in this new discipline. There is a dearth of EM faculty trained in research. "Pub-Lab" was created as a



series of regular meetings where the faculty discuss research articles, bring research ideas, or work in progress for publication. A faculty member trained in research provides guidance, and to optimize mentoring time, "mass-mentoring" groups were formed with each group comprised of faculty who are interested in a subject area/topic. MAD has up to four faculty and a mentor. Published studies from this group mentorship are in Table 69.2.

To Accelerate Knowledge Generation and Synthesis by Bringing Together Clinicians and Scientists from Diverse Disciplines

POEM builds upon existing scientific platforms, including scientific conferences and meetings both at local and national levels. January 2020 was the 7th Oncologic Emergency Medicine Conference at MD Anderson. Sponsored by DEM, among the aims of the conference was to fill the knowledge gap in oncologic emergency medicine by providing information in an interactive forum to enhance the decision-making of healthcare providers. Faculty from diverse disciplines served as speakers including the faculty in DEM. Each year includes presentations on innovative cancer therapies, e.g., immune checkpoint inhibitors, its related side effects and treatment, and management of immune-related adverse events were the highlights for 2020.

To Establish Synergy with Existing Centers and Departments at MD Anderson

As of October 20, 2020, there were 8,128,524 cases of COVID-19 in the United States with more than 218,986 deaths (and rising) [53]. Perhaps, one of the most lethal viruses, SARS-CoV-2, which causes COVID-19, has both an alarming contagion spread and disease severity that often requires emergency care. Our understanding of SARS-CoV-2 infection is rapidly evolving, with its clinical presentation and management barely understood. Cancer patients are expected to be particularly vulnerable to CoVID-19, given that their immune defenses are often deficient or compromised. Early and late toxicities from cancer treatment experienced by cancer patients also add complexity to their care. With little known about COVID-19, research of COVID-19 patients presenting to the ED was quickly established in DEM, and a protocol was immediately developed and obtained approval from the Institutional Review Board. Developing a protocol and a research project allowed for a cohesive method of collaborating with the different centers and departments at MD Anderson and provided synergy and a multidisciplinary approach (radiologists, oncologists, hospitalists, infectious disease) to understanding how to provide care to cancer patients with COVID-19.

Future Directions

Despite the continued rise in the number of cancer patients and the increasing complexity of cancer treatment, there is a dearth of research on oncologic emergencies even for disease complications that have been a concern for decades (acute pain, acute dyspnea, etc.). We have illustrated a couple of research efforts that created opportunities for research collaboration and engaged experts from many disciplines and institutions. Moreover, we also illustrated how the creation of a program in one academic cancer center has facilitated a timely response to research opportunities surrounding the COVID pandemic, for example. In recent years, transformative advances have emerged in the use of innovative therapies for cancer patients (immune therapies, gene therapies, gene editing, etc.). These approaches are some of the most promising treatment modalities but with unknown side effects or complications to the cancer patients. Because many cancer therapies are provided on an outpatient basis, the ED provides a safety net for cancer patients. The provision of care for those who present to the ED has to be guided by evidence.

References

- Miller KD, Nogueira L, Mariotto AB, Rowland JH, Yabroff KR, Alfano CM, et al. Cancer treatment and survivorship statistics, 2019. CA Cancer J Clin. 2019;69(5):363–85.
- National Institutes of Health (NIH). What we do. Budget. https:// www.nih.gov/about-nih/what-we-do/budget. Last reviewed 29 Jun 2020. Accessed 16 Oct 2020.
- National Institutes of Health (NIH). List of NIH Institutes, Centers, and Offices. https://www.nih.gov/institutes-nih/list-nihinstitutes-centers-offices. Last reviewed 19 Aug 2019. Accessed 30 Apr 2020.
- Brown J, Grudzen C, Kyriacou DN, Obermeyer Z, Quest T, Rivera D, et al. The emergency care of patients with cancer: setting the research agenda. Ann Emerg Med. 2016;68(6):706–11.
- Philip J, Remedios C, Breen S, Weiland T, Willenberg L, Boughey M, et al. The experiences of patients with advanced cancer and caregivers presenting to Emergency Departments: a qualitative study. Palliat Med. 2018;32(2):439–46.
- Coyne CJ, Le V, Brennan JJ, Castillo EM, Shatsky RA, Ferran K, et al. Application of the MASCC and CISNE risk-stratification scores to identify low-risk febrile neutropenic patients in the emergency department. Ann Emerg Med. 2017;69(6):755–64.
- Adler D, Abar B, Durham DD, Bastani A, Bernstein SL, Baugh CW, et al. Validation of the Emergency Severity Index (Version 4) for the triage of adult emergency department patients with active cancer. J Emerg Med. 2019;57(3):354–61.
- Caterino JM, Adler D, Durham DD, Yeung SJ, Hudson MF, Bastani A, et al. Analysis of diagnoses, symptoms, medications, and admissions among patients with cancer presenting to emergency departments. JAMA Netw Open. 2019;2(3):e190979.

- Mueller EL, Hall M, Carroll AE, Shah SS, Macy ML. Frequent emergency department utilizers among children with cancer. Pediatr Blood Cancer. 2016;63(5):859–64.
- Richards CT, Gisondi MA, Chang CH, Courtney DM, Engel KG, Emanuel L, Quest T. Palliative care symptom assessment for patients with cancer in the emergency department: validation of the screen for palliative and end-of-life care needs in the emergency department instrument. J Palliat Med. 2011;14(6):757–64.
- Ahn S, Rice TW, Yeung S-CJ, Cooksley T. Comparison of the MASCC and CISNE scores for identifying low-risk neutropenic fever patients: analysis of data from three emergency departments of cancer centers in three continents. Support Care Cancer. 2018;26(5):1465–70.
- Delgado-Guay MO, Rodriguez-Nunez A, Shin SH, Chisholm G, Williams J, Frisbee-Hume S, Bruera E. Characteristics and outcomes of patients with advanced cancer evaluated by a palliative care team at an emergency center. A retrospective study. Support Care Cancer. 2016;24(5):2287–95.
- Banala SR, Yeung SJ, Rice TW, Reyes-Gibby CC, Wu CC, Todd KH, et al. Discharge or admit? Emergency department management of incidental pulmonary embolism in patients with cancer: a retrospective study. Int J Emerg Med. 2017;10(1):19.
- Batalini F, Gomes M, I F, Kuwae F, Macanhan G, Pereira JLB. Cancer complaints: the profile of patients from the emergency department of a Brazilian oncology teaching hospital. F1000Res. 2017;6:1919. https://doi.org/10.12688/f1000research.12632.1. PMID: 30410725; PMCID: PMC6198257
- Baugh CW, Wang TJ, Caterino JM, Baker ON, Brooks GA, Reust AC, Pallin DJ. Emergency department management of patients with febrile neutropenia: guideline concordant or overly aggressive? Acad Emerg Med. 2017;24(1):83–91.
- Brown C, Drosdowsky A, Krishnasamy M. An exploration of medical emergency team intervention at the end of life for people with advanced cancer. Eur J Oncol Nurs. 2017;31:77–83.
- Elsayem AF, Bruera E, Valentine A, Warneke CL, Wood GL, Yeung SJ, et al. Advance directives, hospitalization, and survival among advanced cancer patients with delirium presenting to the emergency department: a prospective study. Oncologist. 2017;22(11):1368–73.
- Elsayem AF, Bruera E, Valentine AD, Warneke CL, Yeung SC, Page VD, et al. Delirium frequency among advanced cancer patients presenting to an emergency department: a prospective, randomized, observational study. Cancer. 2016;122(18):2918–24.
- Elsayem AF, Merriman KW, Gonzalez CE, Yeung SC, Chaftari PS, Reyes-Gibby C, Todd KH. Presenting symptoms in the emergency department as predictors of intensive care unit admissions and hospital mortality in a comprehensive cancer center. J Oncol Pract. 2016;12(5):e554–63.
- 20. Kyeremanteng K, Ismail A, Wan C, Thavorn K, D'Egidio G. Outcomes and cost of patients with terminal cancer admitted to acute care in the final 2 weeks of life: a retrospective chart review. Am J Hosp Palliat Care. 2019;36(11):1020–5.
- 21. El Majzoub I, Qdaisat A, Thein KZ, Win MA, Han MM, Jacobson K, et al. Adverse effects of immune checkpoint therapy in cancer patients visiting the emergency department of a comprehensive cancer center. Ann Emerg Med. 2019;73(1):79–87.
- 22. Mueller EL, Cochrane AR, Lynch DO, Cockrum BP, Wiehe SE. Identifying patient-centered outcomes for children with cancer and their caregivers when they seek care in the emergency department. Pediatr Blood Cancer. 2019;66(10):e27903.
- 23. Peyrony O, Gerlier C, Barla I, Ellouze S, Legay L, Azoulay E, et al. Antibiotic prescribing and outcomes in cancer patients with febrile neutropenia in the emergency department. PLoS One. 2020;15(2):e0229828.
- 24. Reyes-Gibby CC, Melkonian SC, Hanna EY, Yeung SJ, Lu C, Chambers MS, et al. Cohort study of oncologic emer-

gencies in patients with head and neck cancer. Head Neck. 2017;39(6):1195-204.

- Reyes-Gibby CC, Anderson KO, Todd KH. Risk for opioid misuse among emergency department cancer patients. Acad Emerg Med. 2016;23(2):151–8.
- Tanriverdi O, Beydilli H, Yildirim B, Karagoz U. Single center experience on causes of cancer patients visiting the emergency department in Southwest Turkey. Asian Pac J Cancer Prev. 2014;15(2):687–90.
- 27. Yang Z, Qdaisat A, Hu Z, Wagar EA, Reyes-Gibby C, Meng QH, Yeung SC. Cardiac troponin is a predictor of septic shock mortality in cancer patients in an emergency department: a retrospective cohort study. PLoS One. 2016;11(4):e0153492.
- Lamb MJ, Roman E, Howell DA, Kane E, Bagguley T, Burton C, et al. Hodgkin lymphoma detection and survival: findings from the Haematological Malignancy Research Network. BJGP Open. 2019;3(4):bjgpopen19X101668.
- Mayer DK, Travers D, Wyss A, Leak A, Waller A. Why do patients with cancer visit emergency departments? Results of a 2008 population study in North Carolina. J Clin Oncol. 2011;29(19):2683–8.
- Mueller EL, Hall M, Shah SS, August KJ, Radhi MA, Macy ML. Characteristics of children with cancer discharged or admitted from the emergency department. Pediatr Blood Cancer. 2016;63(5):853–8.
- 31. Mueller EL, Jacob SA, Cochrane AR, Carroll AE, Bennett WE. Variation in hospital admission from the emergency department for children with cancer: a pediatric health information system study. Pediatr Blood Cancer. 2020;67(6):e28140.
- Patel PM, Goodman LF, Knepel SA, Miller CC, Azimi A, Phillips G, Gustin JL, et al. Evaluation of emergency department management of opioid-tolerant cancer patients with acute pain. J Pain Symptom Manag. 2017;54(4):501–7.
- Sadik M, Ozlem K, Huseyin M, AliAyberk B, Ahmet S, Ozgur O. Attributes of cancer patients admitted to the emergency department in one year. World J Emerg Med. 2014;5(2):85–90.
- 34. Seow H, Barbera L, Pataky R, Lawson B, O'Leary E, Fassbender K, et al. Does increasing home care nursing reduce emergency department visits at the end of life? A population-based cohort study of cancer decedents. J Pain Symptom Manag. 2016;51(2):204–12.
- 35. Scholer AJ, Mahmoud OM, Ghosh D, Schwartzman J, Farooq M, Cabrera J, et al. Improving cancer patient emergency room utilization: a New Jersey state assessment. Cancer Epidemiol. 2017;51:15–22.
- 36. Qdaisat A, Yeung SJ, Variyam DE, Badugu P, Ghaly F, Rice TW, et al. Evaluation of cancer patients with suspected pulmonary embolism: performance of the American College of Physicians Guideline. J Am Coll Radiol. 2020;17(1 Pt A):22–30.
- 37. Grudzen CR, Richardson LD, Johnson PN, Hu M, Wang B, Ortiz JM, et al. Emergency department-initiated palliative care in advanced cancer: a randomized clinical trial. JAMA Oncol. 2016;2(5):591–8.
- Vandyk AD, Harrison MB, Macartney G, Ross-White A, Stacey D. Emergency department visits for symptoms experienced by oncology patients: a systematic review. Support Care Cancer. 2012;20(8):1589–99.
- Baugh CW, Faridi MK, Mueller EL, Camargo CA, Pallin DJ. Nearuniversal hospitalization of US emergency department patients with cancer and febrile neutropenia. PLoS One. 2019;14(5):e0216835.
- Huang Y-L, Alsaba N, Brookes G, Crilly J. Review article: end-oflife care for older people in the emergency department: a scoping review. Emerg Med Australas EMA. 2020;32(1):7–19.
- 41. Lee YH, Chu D, Yang NP, Chan CL, Cheng SP, Pai JT, Chang NT. Emergency visits among end-of-life cancer patients in Taiwan: a nationwide population-based study. BMC Palliat Care. 2015;14:25.7.

- 42. Mills S, Buchanan D, Guthrie B, Donnan P, Smith B. Factors affecting use of unscheduled care for people with advanced cancer: a retrospective cohort study in Scotland. Br J Gen Pract J R Coll Gen Pract. 2019;69(689):e860–8.
- 43. Panattoni L, Fedorenko C, Greenwood-Hickman MA, Kreizenbeck K, Walker JR, Martins R, et al. Characterizing potentially preventable cancer- and chronic disease-related emergency department use in the year after treatment initiation: a regional study. J Oncol Pract. 2018;14(3):e176–85.
- 44. Rivera DR, Gallicchio L, Brown J, Liu B, Kyriacou DN, Shelburne N. Trends in adult cancer-related emergency department utilization: an analysis of data from the nationwide emergency department sample. JAMA Oncol. 2017;3(10):e172450.
- 45. Tang PL, Chang HT, Cheng CC, Chen HC, Kuo SM, Hsiao KY, Chang KC. An analysis of emergency department visits and the survival rate for colorectal cancer patients: a nationwide populationbased study. Intern Med. 2017;56(16):2125–32.
- 46. Liao CM, Kung PT, Wang YH, Tsai WC. Effects of multidisciplinary team on emergency care for colorectal cancer patients: a nationwide-matched cohort study. Medicine (Baltimore). 2017;96(23):e7092.
- 47. Yap S, Goldsbury D, Yap ML, Yuill S, Rankin N, Weber M, et al. Patterns of care and emergency presentations for people with nonsmall cell lung cancer in New South Wales, Australia: a populationbased study. Lung Cancer. 2018;122:171–9.

- Hryniewicki AT, Wang C, Shatsky RA, Coyne CJ. Management of immune checkpoint inhibitor toxicities: a review and clinical guideline for emergency physicians. J Emerg Med. 2018;55(4):489–502.
- 49. Knight T, Ahn S, Rice TW, Cooksley T. Acute oncology care: a narrative review of the acute management of neutropenic sepsis and immune-related toxicities of checkpoint inhibitors. Eur J Intern Med. 2017;45:59–65.
- Bowers KJ, McAllister KB, Ray M, Heitz C. Ketamine as an adjunct to opioids for acute pain in the emergency department: a randomized controlled trial. Acad Emerg Med. 2017;24(6):676–85.
- 51. Northfield S, Button E, Wyld D, Gavin NC, Nasato G, Yates P. Taking care of our own: a narrative review of cancer care servicesled models of care providing emergent care to patients with cancer. Eur J Oncol Nurs. 2019;40:85–97.
- 52. National Cancer Institute. Division of Cancer Control and Population Sciences. Epidemiology and Genomics Research Program. Comprehensive Oncologic Emergencies Research Network (CONCERN). https://epi.grants.cancer.gov/concern/. Accessed 30 Apr 2020.
- 53. Centers for Disease Control and Prevention. Coronavirus Disease 2019 (COVID-19). CDC COVID Data tracker. Reported to the CDC since 21 Jan 2020; updated 20 Oct 2020. https://covid.cdc. gov/covid-data-tracker/#cases_casesper100klast7days. Accessed 20 Oct 2020.



Knox H. Todd

Check for updates

70

Introduction

As of November 17, 2020, the World Health Organization reports 54,771,888 cases of COVID-19 worldwide and over 1,324,249 deaths [1]. The cumulative death total in the USA is almost one-quarter million [2]. Despite the promise of an effective vaccine in 2021, new cases and deaths may double in the coming months [3]. In addition to this toll, livelihoods have been destroyed, poverty rates and food insecurity are rising, and even our ability to gather with loved ones is curtailed.

Sadly, the USA has fared much worse than other developed countries during the pandemic. Basic public health measures to contain the COVID-19 epidemic (e.g., social distancing and mask wearing) have been interpreted as political acts in a highly polarized country. It is hoped that scientific reason will inform future developments in the USA, not the least to protect the most vulnerable among us, including those with cancer.

Longstanding inadequacies of our public healthcare systems and declines in longevity and well-being among US citizens have become more glaringly obvious during the pandemic, as has evidence of racial/ethnic disparities, driven by structural racism and workplace exposure to SARS-CoV-2 [4, 5].

In the midst of the pandemic, with rapid and ongoing changes in recommendations for management and the likelihood that a vaccine is forthcoming, it is difficult to discuss treatment issues with confidence in a traditional textbook. Alternatively, readers are directed to the large number of upto-date online resources for guidance on current evaluation and clinical management (e.g., CDC, WHO, etc.). This chapter will briefly review real and potential impacts of the COVID-19 pandemic on the emerging field of oncologic emergency medicine. Among other questions, we will review COVID-related risk factors and health disparities, impact on

Department of Emergency Medicine, MD Anderson Cancer Center, Houston, TX, USA treatment and cancer care systems (including cancer prevention, research, and economic effects), workforce mental health issues, and advances in emergency department (ED) palliative care.

Epidemiology

Cancer patients are at high risk for severe COVID-19 for a number of reasons. Age, obesity, and smoking are common risk factors for both cancer and COVID-19 morbidity and mortality. Advanced cancer causes declines in functional status (another risk for poor outcomes) and anticancer therapies alter immune responses to viral infection [6].

Giesen et al. recently reviewed risk factors for severe COVID-19 disease among cancer patients [7]. Patient-related factors included higher age, male sex, higher ECOG scores, the number of comorbidities, and smoking [8–14]. Cancer-related risk factors included a history of cancer [15], hematologic malignancies vs. solid tumors [10, 11, 16–21], lung cancer vs. other solid tumors [22], active cancer [9, 17], metastatic cancer [8, 22], and cancer treatment within 4 weeks of COVID-19 disease onset [23–25]. In addition, lymphopenia and granulocytosis were associated with an increased risk of severe disease or death [8, 13, 18, 25–27]. Whether cancer is an independent risk factor for poor outcomes in patients with COVID-19 is unclear at this time.

Initial reports from China by Liang et al. suggested that COVID-19 patients with cancer had higher rates of ICU admission and death; however, at the time of their publication, the investigators had evaluated only 18 patients with cancer from a cohort of 1590 subjects [28]. A subsequent study using propensity matching compared 232 adult cancer patients to 519 non-cancer controls admitted with COVID-19 to 9 area hospitals in Wuhan, China [8]. Over an approximately 1-month follow-up period, those with cancer were more likely to experience severe disease (64% vs. 32%) and death (20% vs. 11%). Risk for disease severity and death were greatest for patients receiving recent chemotherapy

© Springer Nature Switzerland AG 2021

K. H. Todd (\boxtimes) (Retired)

K. H. Todd et al. (eds.), Oncologic Emergency Medicine, https://doi.org/10.1007/978-3-030-67123-5_70

(within the past 2 weeks). In addition to known clinical risk factors for poor outcomes, the investigators identified a number of novel risk factors for disease severity among cancer patients, including advanced tumor stage, elevated tumor necrosis factor alpha (TNF- α) and N-terminal pro-B-type natriuretic peptide (NT-proBNP), and decreases in CD4+ T cells and albumin-globulin ratios. They recommended heightened attention to infection control procedures during treatments, considering delays of adjuvant chemotherapy or elective surgery in endemic areas (with continuation of oral medications), and intensive risk stratification for patients with cancer and COVID-19, using the above identified biomarkers.

A study of 105 cancer patients and 536 without cancer admitted to 14 hospitals in Hubei Province, China, reported a higher risk for multiple poor outcomes (i.e., disease severity, ICU admission, mechanical ventilation, and death) among those with cancer [22]. Patients with hematologic, lung, or metastatic malignancies were at higher risk, as were those undergoing surgery vs. receiving only radiotherapy.

A more recent study performed COVID-19 surveillance among 1081 consecutive patients with solid tumors receiving treatment with intravenous, subcutaneous, or intramuscular agents at the National Cancer Institute of Milan (The COVINT Study) [29]. They found that over 2 months of epidemic spread in Milan, only 11 patients (1%) had confirmed, and 73 (6.7%) suspected, COVID-19. They concluded that when protective measures were followed in administering chemotherapy, COVID-19 contraction rates in cancer treatment facilities were relatively low.

A subsequent analysis comes from the LEOSS (Lean European Open Survey on SARS-CoV-2-Infected Patients) Study registry, in which a retrospective cohort of 435 cancer and 2636 non-cancer patients were assessed for COVID-19 morbidity and mortality [30]. For 435 cancer patients identified at the time of COVID-19 detection, progression to severe disease was seen in 55% and ICU admission in 27.5%, with a COVID-19-related mortality rate of 22.5%. Male sex, advanced age, and active malignancy were associated with higher mortality. After adjusting for confounding factors, mortality rates for those with and without cancer were similar. Despite this finding, cancer patients *as a group* are at higher risk of poor outcomes, as they are older and tend to have multiple comorbidities.

Analyses of UK Coronavirus Cancer Monitoring Project data provide more specificity for cancer as a risk factor for poor outcome. Investigators found that all-cause case fatality rates for patients with cancer who were hospitalized for COVID-19 rose with increasing age and the presence of hematologic malignancies [21]. After controlling for age and sex, those with hematologic malignancies receiving recent chemotherapy had an approximate doubling of mortality risk (OR 2.09, 95% CI 1.09–4.08).

Health Disparities

For the USA, COVID-19 represents the latest chapter in the long and tragic history of national health disparities. The pandemic has focused attention on persistent inequities experienced by US subpopulations facing persistent and collective discrimination (e.g., specific racial/ethnic groups and those of low socioeconomic status). As is true historically for other communicable and non-communicable diseases (e.g., tuberculosis, pellagra, HIV), to address the COVID-19 pandemic, we must understand its differential impact on these groups and focus our efforts to ensure more equitable monitoring and treatment, incorporating what we already know about the social determinants of health [31].

Current observational studies provide abundant evidence of disparities in both risk and outcomes for COVID-19. In retrospective cohort studies within a Louisiana integrateddelivery health system, researchers found that while Blacks comprised only 31% of those they served, 77% of those hospitalized were Black, as were 71% of those who died [32]. The authors note that in addition to a higher incidence of COVID-19 risk factors (e.g., obesity) among Blacks, most service workers in New Orleans and surrounding areas are members of minority groups, with many service sector occupations related to food preparation and serving.

Similarly, an analysis of data from Sutter Health, an integrated California healthcare system, found that among 1052 confirmed COVID-19 cases, after adjusting for age, sex, comorbidities, and income, Blacks were 2.7 times as likely as Whites to be hospitalized [33]. The authors posited that Blacks were more likely to be tested at a later stage of illness, as evidenced by their testing occurring more often in hospital EDs than in office or clinic settings.

In an observational cohort study of 2186 US adults with invasive cancer and COVID-19, Rivera et al. found that while remdesivir was associated with improved outcomes, Black patients were approximately one half as likely to receive this therapeutic agent as Whites [34]. Of note, remdesivir is most often administered within a clinical trial. Although an observational study, these findings suggest that racial inequity, particularly as related to clinical trial enrollment, continues to influence COVID-19 treatment.

Using publicly available datasets, Tirupathi et al. reported that Black and Latinx COVID-19 incident rates were disproportionately higher than their population percentage in 14 states and 9 states, respectively, and that while Blacks constitute 13.4% of the US population, they accounted for 22.4% of COVID-19 deaths [35].

Geographic disparities are also important to consider in the context of the COVID-19 pandemic. Initial waves of the virus impacted predominantly urban areas; however, the disease is now widely distributed in the USA, with less densely populated parts of the country experiencing disproportionate levels of hospitalization and death. Rural health disparities result from multiple factors: poor access to healthcare (e.g., higher levels of the uninsured, long travel distances, lack of public transportation, poor Internet connections, lack of specialty care) and lower socioeconomic status (e.g., higher unemployment, lower median household incomes, lower proportions with post-secondary education), among other factors.

Dr. Robert Rodriguez, an emergency physician recently named to President-Elect Biden's COVID-19 Advisory Board, recently wrote of his experiences in volunteering to care for critically ill patients in his hometown of Brownsville, Texas, contrasting it to his current work environment in San Francisco: "My experiences over the past 2 months have shed light on another disparity—the enormous differential in emergency and critical care physician surge capacity between cities with robust medical academic institutions and communities that are situated far away from large medical schools" [36]. As part of his critique, he noted the bureaucratic barriers (e.g., credentialing paperwork, malpractice insurance) to urban physicians volunteering to serve in rural COVID-19 hot zones.

In rural and urban areas, for many of those with lowpaying service jobs, social distancing or telecommuting are not options, and adverse social determinants of health are a large factor in observed COVID-19 health disparities [37]. This poor state of affairs is exemplified by the title of a recent JAMA editorial discussing COVID-19: "Failing Another National Stress Test on Health Disparities" [38]. While health disparities were widely recognized before COVID-19, the pandemic has served to highlight their importance and further focus our attention on this ongoing national tragedy.

Treatment

As noted in the Introduction, we will cover treatment issues in only a cursory fashion. Given the time interval required for publication, recommendations made today are likely to be misleading by the time this text is available. As a recent review of COVID-19 emergency medicine response noted: "When we are practicing at the bleeding edge of a viral pandemic that didn't exist 6 months ago, practitioners are often forced to work with less than robust data sets" [39]. Our readers are best referred to online sources from emergency medicine, oncologic, and federal/state organizations for the latest updates on evaluation and treatment.

In terms of management lessons learned during the pandemic, EDs have met multiple challenges involving isolation protocols and staff safety, patient/provider education, public health reporting, staff and space allocation, and communications – not to mention the widely publicized (real and potential) shortages of personal protective equipment and ventilators – as well as varying policy recommendations from federal, state, and professional authorities. Screening and testing criteria also evolved over time, as did criteria to justify hospital admission.

However, EDs, in part due to past experience with previous mass casualty incidents and epidemics, adapted rapidly to enhance surge capacity. They established separate "hot" and "cold" zones for COVID and non-COVID care, respectively, and in some cases, utilized separate tent and drive-in facilities to further segment care and augment capacity. In particularly stressed areas, adaptions could be primitive (e.g., using bandannas or handkerchiefs and plastic trash bags as a substitute for sophisticated personal protective equipment) [40]. Social media-driven efforts were conducted to raise funds for PPE in order to ameliorate inequalities in supply distribution As voiced by one emergency physician, "I think it should not have required a social media presence of health care workers, and for health care workers to get sick and themselves die from coronavirus, in order for hospitals to have received the standard equipment that they deserve" [40].

One of the most difficult decisions, and one for which lessons continue to be learned, surrounds questions of oxygenation and intubation. The risk of spreading viral-laden droplets and aerosols during oxygenation and intubation procedures complicates decision-making. In addition to standard protective equipment, powered air-purifying respirators (PAPR) are commonly used to reduce these exposures [41]. With more vigorous non-invasive oxygenation efforts, patient positioning techniques ("proning") [42], and recognition that tolerance of low patient oxygen saturation levels ("happy hypoxemia") can occur [43], intubation procedures are often delayed. ED airway management has improved markedly over the past decades and the COVID-19 experience is likely to further improve and standardize our airway practices [44]. Figure 70.1 presents an algorithm from NorthShore University HealthSystem intended to reduce aerosol exposure risk and maximize alternative oxygenation strategies for COVID-19 [39].

Pharmacologic approaches to both ED and oncologic patients continue to evolve. Beyond remdesivir, corticosteroids, and anticoagulants, the potential roles of convalescent plasma, monoclonal antibodies, antivirals, interleukin and cytokine blockers, histamine antagonists, kinase inhibitors, and protease antagonists in COVID-related disease management remain unclear [45]. The ED's place in future vaccine distribution systems over the coming year(s) is, as well, unclear; however, there is no doubt that emergency care providers will be among the first recipients of any new vaccines.

Cancer centers have also adapted rapidly to the challenges of treating vulnerable and potentially immunocompromised patients in the midst of a viral pandemic. Broom et al.

COVID-19 respiratory distress algorithm

Work of breathing

(Respiratory intervention should be based on work of breathing, NOT merely pulseOx readings)



Fig. 70.1 Example COVID-19 resiratory distress algorithm. (From Leiker and Wise [39] with permission Elsevier)

recently provided an excellent review of modifications in cancer care resulting from COVID-19 [46]. In their review, they recognize that rapidly implemented responses to the pandemic may result in enduring changes to the cancer care system (for both better and worse). They touch on key principles to consider in understanding and managing such broad changes, including: (1) recognition that a return to "business as usual" after the pandemic is unlikely, (2) safety for patients and staff is a priority, (3) the pandemic requires a heightened focus on the negative consequences of immune-altering therapies, (4) the need for rapid evaluation of changes in practice, (5) ramifications (both interpersonal and economic) of increased telemedicine use, and (6) increased scrutiny of value of life assessments in the context of limited resources, particularly for the elderly, patients with palliative care needs, and those with multiple comorbidities.

Percival et al. reviewed early modifications of care for particularly fragile patients with hematologic malignancies served by the Fred Hutchinson Cancer Consortium [47]. Their overarching goal was to balance the risks surrounding the underlying hematologic malignancy and its treatment (often involving myelotoxic and lymphotoxic therapies) with evolving understanding of COVID-19 morbidity and mortality.

Aside from stringent infection control measures (i.e., strict control of access to healthcare sites, increased use of telemedicine, limiting the physical presence of caregivers), they increased their emphasis on oral and/or outpatient options, choosing therapies that reduce the risk for bone marrow toxicity and deferring therapy, if possible. There was also a marked reduction in clinical trial participation and a more nuanced calculation of risk-benefit for experimental therapies. Increased use of granulocyte colony-stimulating factor (G-CSF) and antibiotic prophylaxis was emphasized in an attempt to limit febrile neutropenia, as were stringent transfusion thresholds (given decreased donor availability), increased use of anti-fibrinolytics to decrease risks of spontaneous bleeding in thrombocytopenic patients, and delayed therapy for patients with positive coronavirus tests, when possible.

Given limited testing capacities, they proposed screening asymptomatic patients for coronavirus only prior to planned procedures, inpatient admission, stem cell transplant (HSCT), and chimeric antigen receptor (CAR)-T cell therapy. Finally, as in the ED, they increased attention to early discussions of patient and family goals of care, encouraging frank assessments of risk for those with limited expectations of benefit from more aggressive therapies. The ongoing reevaluation of cancer regimens in the COVID-19 era, with potential de-escalation of therapy when appropriate, holds the promise of more cost-effective, patient-centered care in the future [48]. A summary of general consensus measures [48] taken in response to the pandemic by seven comprehensive cancer centers in Europe is presented in Table 70.1. These measures are specific to a particular point in time in the evolution (as well as our understanding) of the pandemic and they will, no doubt, change over time.

Throughout the healthcare system, efforts to respond to the challenges of COVID-19, particularly in the context of cancer care needs, have had both positive and negative consequences. In prioritizing our response to the epidemic, certain aspects of cancer care have necessarily been deprioritized, occasionally resulting in suboptimal or delayed care.

Indeed, medical lawsuits in Spain may have resulted from delayed diagnosis and treatment of cancer during the pandemic [49]. The story noted that while hospitals were focused on COVID-19 treatment, cancer and other serious illnesses might have been neglected. While Spanish prosecutors are investigating such cases, physicians interviewed for the article stated that their overwhelming workload makes such errors inevitable, and specifically that when forced to rely on telemedicine only to evaluate patients, their ability to detect cancer will necessarily suffer.

In April of 2020, Denise Grady of the *New York Times* reported similar instances of COVID-19 collateral damage, in the story of a 53-year-old male with a hematologic malignancy whose death may have been caused by delays in chemotherapy due to COVID-related shortages of blood products. The article cited an American Cancer Society Cancer Action Network survey conducted at the beginning of the pandemic reporting that one-half of cancer patients and survivors experienced some impact of the pandemic on their care [50]. The survey found that 27% of patients in active cancer treatment report treatment delays, with 13% reporting no knowledge of when treatment might be rescheduled. Fully 40% of those in active treatment expressed concern regarding their ability to receive adequate care.

System Impacts

Healthcare system stresses (e.g., lack of surge capacity) and altered patient behaviors related to the pandemic (e.g., avoiding ED and office visits) are likely to lead to delayed cancer diagnoses at later stages and with poorer outcomes. Early data in the pandemic era reveal dramatic drops in cancerrelated patient encounters. London et al. examined patient encounter data from 21 healthcare institutions (20 from the USA and 1 from the UK) and reported a 57% decline in US cancer-related visits and a 50% decrease in UK visits due to the pandemic [51]. Patient visits for new cancer incidence declined by 74% and 65%, respectively. They also report profound decreases in screening for breast (89%) and

Table 70.1 Ge	eneral consensus measures taken by cancer centers in response to COVID-19
Category	Measure
Hospital wide	Construct a hospital-wide crisis team responsible for coordinating measures between departments
	Encourage patients not to arrive early. Offer to text patients when you are ready to see them, so they can wait outside or in the car
	Instruct patients not to visit the hospital if they have symptoms indicative of possible COVID-19 (unless urgent attention is required)
	Call patients the day before planned hospital admissions, to discuss the presence of any COVID-19-related symptoms
	Screen patients at the entrance for symptoms of COVID-19 and fever
	Quickly isolate patients with COVID-19 in specialized departments, with the intent of relocation to regional collaborating hospitals (if possible)
	Reduce preclinical research activities to a bare minimum
	Stop patient inclusion for clinical studies or trials requiring additional actions and/or visits. Consider a tumor type-specific "exception list" of particularly successful studies for which inclusion continues
	Discuss each patient with a multidisciplinary team to consider alternative treatment modalities with the fewest visits or lowest capacity problems or that are the shortest in duration
	Therapeutic adjustments (versus regular guidelines) should be discussed in a multidisciplinary team meeting
	Conduct multidisciplinary team consultations remotely if possible or include only one representative of each discipline to limit the number of people participating in the meetings
	Inform patients about a possibly increased risk associated with anticancer therapy during the COVID-19 pandemic
	Enable telephone or video consultations for healthcare professionals who need to self-isolate
	When postponing procedures or contact moments, anticipate future capacity problems
	Do not prescribe corticosteroids as anti-emetics (if avoidable), and limit their use in patients treated with immune-checkpoint blockade, to reduce vulnerability to COVID-19
	With each patient, discuss resuscitation status to anticipate future decisions about intensive care
Outpatient	Critically triage second opinions
clinic	Do all follow-up appointments by phone (except when physical examination is necessary)
	When possible, reduce or delay the number of radiological-response evaluations
	Prioritize oral or subcutaneous treatments above infusion-based treatments to reduce time spent in the hospital
	Perform blood tests outside the hospital (e.g., at a general practice or at home), when possible
	Have oral medications delivered to the patient's home, rather than being picked up at the pharmacy
Day care	Consider omitting supportive treatments (e.g., no bisphosphonate infusion, except in the case of hypercalcemia)
	When possible, organize the administration of intravenous maintenance treatments at home
	When administration at home is impossible, consider temporary breaks or reductions in the frequency of intravenous maintenance treatments for less-aggressive metastatic cancers on a per-patient basis
Radiotherapy	Consider hypofractionated regimens for patients with limited additional benefit of regular regimens
	Create capacity for radiation as replacement of surgery
Surgery	Consider postponement of surgeries with high morbidity and mortality during the pandemic
	Consider other treatment modalities with equal benefit (e.g., radiation for prostate cancer, curative chemoradiation for other tumor types, or brain irradiation for metastases)
Others	Consider outsourcing of interventions (e.g., follow-up endoscopies) to private clinics

From van de Haar [48], with permission Springer Nature

colorectal cancer (85%). The impact of deferring cancer treatment and screening is difficult to estimate, and it is unclear how long these visit levels will remain depressed; however, it seems likely that we will see future increases in delayed cancer diagnoses and later-stage cancer presentations to the ED.

With regard to trends in radiology use, Norbash et al. examined imaging volumes in six US academic medical systems and a large national private practice consortium. As expected, they reported large decreases (40–70%) during early stages of the pandemic, with the largest decreases occurring in screening mammography and bone density scanning [52]. Interestingly, in discussing a pandemicinduced increase in telecommuting by radiologists, they noted the potential for a new source of tension between frontline and remote healthcare workers, stating that: "An unfavorable byproduct of radiologist distancing at work and by telecommuting may be the impression of radiologists utilizing technologists, nurses, and residents as human shields while maximizing radiologists' physical perimeter. The optics are potentially damaging in the long term for the larger radiology team beyond radiologists and may counteract loyalty and high-performance interdependence" [52].

This sentiment above reflects a possible negative consequence of the move toward telemedicine; however, COVID-19 continues to foster the dramatic growth of such virtual care. In part, this impetus derives from reductions in both practice regulations and reimbursement barriers [53]. As part of the US federal stimulus package passed in March of 2020 (The Coronavirus Preparedness and Response Supplemental Appropriations Act or CPRSA), rules surrounding reimbursement for telemedicine were substantially relaxed [54]. Prior to CPRSA, CMS only reimbursed telehealth services when patients received care in urban or rural locations with a known health professional shortage. Recipients must also have had a prior established relationship with the physician, a limited number of services were covered, and only certain HIPAA-compliant platforms were allowed. After CPRSA, these requirements were loosened and CMS specified that telehealth services would be reimbursed at the same levels as in-person services. Private insurers generally followed suit, thus the use of telemedicine skyrocketed.

While the virtual delivery of medical care has many positives, there are, of course, negative consequences. In physically distancing ourselves from patients, we may also distance ourselves emotionally [46]. We have discussed how an inadequate assessment might lead to failures of diagnosis, but there are also failures of rapport- and trust-building between patients and clinicians that might be increased by use of virtual interfaces. Geographic and socioeconomic inequities in Internet connectivity may also serve to increase health disparities. It is possible that the popularity of telemedicine among providers and patients will sustain its current levels of use; however, it is also possible that future reductions in reimbursement, in addition to anti-competitive medical market forces, will counterbalance this trend.

With regard to cancer research, the COVID-19 pandemic has caused profound disruption. Most clinical trials were suspended, while the research enterprise has shifted its focus to pandemic-related themes. Many NIH-supported research programs were allowed latitude to shift in scope toward COVID-19 research with federal encouragement [55]. At the same time, there has been a drastic reduction in cancer-related philanthropy. As only one example, the American Cancer Society projects a \$200 million (25%) decrease in donations over 2020 [56].

This disruption in the research enterprise does have a potential upside. As cancer research efforts pivot toward COVID-19, they may allow a more precise assessment of treatment toxicities and outcomes. As van de Haar and colleagues have outlined, four research priorities should drive current efforts: (1) to collect real-world data on how adjustments and de-escalation of treatment impact patients; (2) assess symptomatic and asymptomatic COVID-19 among those receiving chemotherapies, targeted therapies, or immune-checkpoint inhibitors; (3) develop epidemiologic models to estimate the cumulative incidence of COVID-19 among those with cancer within a specific timeframe; and (4) estimate COVID-19 morbidity and mortality in patients with cancer who are treated with chemotherapy, targeted therapy, immune-checkpoint blockade, and/or G-CSF [48]. COVID-19 thus provides a unique opportunity to focus on

the impact of treatment de-escalation, which for ethical reasons might otherwise be difficult.

Prevention

As noted above, the pandemic has caused large shifts in our ability to maintain current cancer care practices. At the national level, both the Welsh and Scottish governments suspended breast, cervical, and colon cancer in March of 2020, while Northern Ireland suspended screenings 1 month later [57]. In the USA, the Centers for Medicare and Medicaid Services lowered the priority for cancer screening programs at the outbreak of the pandemic [58]. Patient fear of exposure to the virus also served to decrease rates of cancer screening.

Tobacco prevention programs have experienced a similar COVID-related deterioration. Tobacco consumption has likely risen during the pandemic at the same time that access to healthcare has been disrupted. It is thus likely that tobacco-related morbidity and mortality will rise over the coming months and years [59]. Smokers are of particular interest, as the pandemic may both increase smoking behaviors in some (due to boredom and lack of social contacts during quarantine) and decrease such behaviors in others (due to a desire to maintain health in the face of external threats) [60].

Smoking may also impair the ability to utilize masks effectively (envision a mask dangling around the neck surrounded by cigarette smoke plumes) and increase mucous membrane viral exposure through contaminated fingers, while encouraging smokers to be outdoors (perhaps thereby decreasing exposure risk). Fewer visits to smoking cessation resources may mean a drop in use of nicotine patches and smoking cessation medications. Compounding these issues, medical care for chronic smoking-related conditions has been curtailed. Consistent with these expectations, EDs have indeed reported fewer admissions and delayed admissions for myocardial infarction and stroke since the pandemic began [61].

Surprisingly, initial reports suggested that smoking might have a protective effect against COVID-19, given that the prevalence of current smokers among hospitalized patients in China was unexpectedly low [62]. A cross-sectional analysis of patients presenting in New York City with COVID-19 found that smoking was not a risk factor for critical illness, and that it might even serve as a protective factor for hospital admission [63].

As another example of screening trends, Murray et al. report large reductions (as much as 58%) in referrals to Irish dermatologists for evaluation of pigmented skin lesions at the height of the pandemic [64] (Fig. 70.2). Interestingly, they also reported decreases in Internet searches (as a proportion of total searches) for both skin cancer and melanoma,

NCCP pigmented lesion referrals in Ireland



Fig. 70.2 National pigmented lesion referrals recorded from the NCCP Ireland during 2019 and 2020 of the months of January to June. (From Murray et al. with permission John Wiley)

as COVID-19 search activity swamped all other terms. Earnshaw et al. note similar declines in skin cancer referral (as much as 56%) [65]. They also make the valuable point that data regarding declines in cancer detection rates will assist in planning for the inevitable increase in diagnoses (often at later stages of disease) that will occur during any eventual pandemic recovery stage.

Economic Considerations

During the early months of the pandemic, while the numbers of COVID-19 visits to US EDs increased, the total number of ED visits plunged dramatically. Figure 70.3 illustrates this fall in patient volume for both academic and community EDs in one Massachusetts system [66] and others report similar findings [67, 68]. Data from the National Syndromic Surveillance Program show that national ED visits declined by over 40% during the early months of the pandemic, with the steepest declines among children and females and in the Northeast [69].

The public avoided the ED if possible, given the graphic images of ED chaos depicted in the media, the fear of contracting the virus, and a desire to avoid overburdening the system. In addition, while the public remained at home, injury rates outside the home (e.g., traffic injuries) decreased and fewer communicable diseases (other than COVID-19) required emergent care.

With this decrease in ED visits, the number and proportion of lower-acuity presentations fell, as did the need for a number of non-critical (and highly reimbursed) ED procedures. Most subspecialties experienced a fall in ED consulta-



Fig. 70.3 Daily number of ED visits over time across the five included hospitals, separated by academic and community hospitals. (From Baugh et al. [66], with permission Elsevier)

tions, as well. Given that many hospitals cancelled elective surgeries and other procedures, the rate of ED visits for postprocedural complications declined in a similar fashion.

Case Study

Your sister, a 58-year-old breast cancer survivor and her spouse decided to take a rafting trip with two other couples. Before traveling, they decided to obtain tests for coronavirus and drove to a nearby free-standing ED where their nasopharyngeal samples were taken. The results were subsequently negative and they very much enjoyed the trip with friends.

Your sister's spouse paid \$150 in cash for testing while your sister gave the clerk her insurance card. Two weeks after returning from vacation, she received a bill for \$5000, leaving her responsible for substantial charges that the insurer did not cover. She leaves a message for you with her story and asking how such a discrepancy in charges is possible. How do you explain this?

Fortunately, you remember that the New York Times has published a number of articles by Sarah Kliff on the issue of surprise medical billing [70, 71]. Surprise billing occurs when billing services used by "out-of-network" emergency physicians or other clinicians seek payment directly from patients when there is a gap in coverage. Patients often have no knowledge (or way of knowing) that this might occur when seeking care in the midst of an emergency. The issue has been festering for years without resolution; however, given increased attention to billing issues during the pandemic as well as the economic crisis, it is likely to be resolved by legislation that will ultimately decrease payments to physicians and EDs.

After a quick review, you understand that wide variations exist for similar services, even in the same geographic area. Variations exist for a number of reasons, but the underlying reason is that the government does not regulate healthcare prices. Each facility negotiates with a number of insurers, and patients served by smaller insurers tend to pay larger bills. Those patients whose insurers have not negotiated with the charging entity may be responsible for the entire amount of the charge and these fees can be enormous. This can occur when a patient is served by an out-of-network provider, either the entire facility or an individual physician (e.g., emergency physician, anesthesiologist), who is providing services as an independent contractor and charging separately.

In coping with surprise medical billing for coronavirus testing, Sarah Kliff offers a number of recommendations [71]: (1) obtain your test at a public site (e.g., state, county, or city-operated), (2) use your primary care physician or a federally qualified health clinic, (3) avoid hospitals and freestanding EDs (these locations charge facility fees, which are substantial).

While ED visit rates have more recently waxed and waned with pandemic trends, the impact on emergency physician and hospital finances has been overwhelmingly negative. Telemedicine use has expanded, replacing the need for many ED and office visits, and thus far, regulation of telehealth continues to be relaxed with many restrictions on reimbursement lifted. It is difficult to know whether these trends will continue, but to the extent they substitute for lower acuity, privately insured, and more profitable ED visits, emergency physician compensation is likely to fall [72]. The trend toward fewer high-margin ED visits may portend less need for emergency physicians and ED closures in low-margin settings, both urban and rural.

In its long-running biannual national survey of US physicians, the Physicians Foundation focused exclusively on the pandemic [73]. In the first of its three-part survey, conducted from July 15-26, 2020, they report that 8% of physicians (estimated to represent approximately 16,000 medical practices) had closed their practices permanently because of the epidemic and 43% have reduced their office staff, with 72% reporting loss of income and 59% expecting that COVID-19 would lead to a reduction in the number of independent practices in their communities. Similarly, a September 2020 survey of primary care practitioners found that 54% have furloughed employees and 28% have permanently reduced staff; 19% report that clinicians have retired early; and 15% state they are leaving (or planning to leave) their practice. Such losses in community-based medical practices will only increase stresses on ED capacity.

Cancer centers, particularly larger centers with a national or international clientele, have experienced large declines in service demand and face large pandemic-related economic losses. As in other parts of the healthcare system, the declines are in patient encounters that tend to be more profitable, as care moves out of the hospital and elective procedures are postponed or cancelled. In addition to patient fears of viral exposure, many patients who would seek cancer care nationally or internationally are electing to remain near their homes for treatment. Parking garages (a large source of income for many centers), restaurants, hotels, rental services, and the travel industry at large are all experiencing losses related to cancer center business turndowns.

The pace of economic recovery at these centers after the pandemic is an open question. It seems likely that cancer patients will be less willing to travel such great distances if care is available locally, and cancer treatment services have become more dispersed geographically over time. When seeking cancer care out of state, costs for travel, lodging, and lost wages alone are estimated to be double that for in-state care [74]. The disruption and suspension of many clinical trials during the pandemic will also lead to economic losses that may be slow to recover.

Workforce Mental Health

Emergency medicine practice is stressful. Our specialty has long recognized the intense nature of our practice and the challenges it poses to its practitioners. Among medical specialists, emergency physicians experience one of the highest rates of burnout. Emergency physicians suffer from practicerelated anxiety, depression, and depersonalization with some frequency [75].

COVID-19 has only heightened these issues, particularly in areas of early pandemic spread. Studies of Chinese healthcare workers found high rates of mental distress for those involved in the COVID-19 response; moreover, emergency providers experienced the greatest risk for multiple negative outcomes [76]. During the pandemic, particularly in the New York area, emergency physicians faced staff, space, and personal protective equipment shortages. Clinical practice was conducted in an atmosphere of crisis, and emergency physicians often found themselves treating friends and colleagues. Initial reports estimated a 46% emergency physician seroconversion rate (admittedly using a limited sample of physicians from at a single New York ED) [77]. A deluge of news from around the world heightened a sense of emotional and social crisis. Physicians often responded to institutional mandates that were delivered without a sense of consent. Contradictory guidelines regarding PPE were a particular source of anxiety. Would I contact the disease? Would I infect my family? Not surprisingly, it is predicted that our specialty will experience an increase in anxiety disorders and PTSD as a result of COVID-19 [78].

To better quantify COVID-related stress among emergency physicians, Rodriguez et al. conducted a crosssectional survey among academic emergency physicians from seven EDs [79]. The 426 respondents reported high levels of anxiety for themselves and their families. In terms of measures to relieve this anxiety, they desired more widely available PPE, rapid coronavirus testing, and clear communication regarding ED protocols related to COVID. Providing more specificity, Wong et al. outlined a number of practical initiatives, both individual and administrative, to combat mental distress within the emergency medicine workforce (Table 70.2) [80].
 Table 70.2
 Potential solutions for COVID-19-related emergency provider stressors, ordered by Maslow's hierarchy of needs

Maslow's level of need	COVID-19 concerns	Recommended strategies
Level 1: Physiologic (e.g., food, sleep, physical and mental health)	Extra workload demands around COVID-19 preparation and treatment Physical strain of protective equipment (dehydration, heat, exhaustion) Housing needs during isolation/quarantine periods Inadequate or disrupted sleep patterns Physical symptoms of COVID-19 disease for healthcare workers who contract the virus	Individual Time for basic bodily care and refreshment/relaxation and stress-management breaks Avoid maladaptive behaviors with negative physiologic effects (e.g., excessive alcohol, prescription drugs) Physical health and fitness (exercise programs, walking outside, mobile applications) Online mental health technologies (telepsychiatry, mobile applications, PTSD coach) Administrative Provision of respite for staff members requiring isolation (e.g., housing, childcare) Supplementation of readily available water and nutritious food while on clinical duty Careful attention to individual work schedules to maximize rest and sleep between shifts Facilitation of testing and treatment for individuals who develop symptoms or become ill Virtual wellness and information town halls Early and confidential recognition, detection, and referral for treatment of psychiatric symptoms (e.g., cognitive- behavioral therany)
<i>Level 2: Safety</i> (e.g., personal security, financial security, resources)	Fears of personal safety around infection and lack of adequate personal protective equipment Lack of clarity around viral transmissibility (airborne versus droplet) Concerns for job security and potential debt, especially if an individual becomes infected with COVID-19 Feelings of being undersupported and underequipped to provide safe care	Individual Peer consultation and supervision of PPE donning/doffing Administrative Alternative strategies to produce/distribute PPE (local manufacturers, donations, recycling) Clear and consistent messaging and shared decision- making with healthcare workers regarding infection rates, risk, and strategies to minimize risk Contingency plans for healthcare workers who cannot work during quarantine period or if they fall ill after contracting COVID-19 to provide job and financial security without negative consequences
<i>Level 3: Love and belonging</i> (e.g., friendship, family, social connectedness)	Possible separation from family members Risk of exposure to loved ones, especially those who are at high risk Physical isolation from friends, colleagues	Individual Increase peer social support with regular contact with colleagues, family, and friends Seek out and share social support virtually Administrative Acknowledgment and affirmation of healthcare worker stressors and concerns Creation of specialized collaborative partnerships or teams focusing on COVID-19 Online-based group support networks and mental health checks Resources for significant others and family members of healthcare workers to support their loved ones during epidemic
<i>Level 4: Esteem</i> (e.g., respect, status, self-determination/control, fairness)	Pressure to serve as source of definitive information for nonmedical family and friends Constant pressure to maintain clinical acumen with increasing volume and acuity Ethical challenges in triaging resources (ventilators, staffing, bed capacity)	Individual Limit worries to actual (rather than anticipatory) threats Foster a spirit of patience, fortitude, tolerance, and hope Channel concerns through productive output (scholarly efforts, peer coaching, teaching, educational materials on COVID-19) Administrative Create specialized ethics teams/protocols for information and mentorship in decision-making Use patient-centered resources for difficult decisions Highlight exemplary behavior and celebrate individual contributions and efforts Create clear, transparent, fair, equitable, and accessible policies

(continued)

Table 70.2 (continued)

	COVID 10	D 11.4.4
Maslow's level of need	COVID-19 concerns	Recommended strategies
Level 5: Self-actualization (e.g.,	Tension between public health priorities and	Individual
desire for higher achievement)	individual patient care	Focus on efforts within one's individual control
	Advocacy for larger system changes to	Accept situations one cannot change
	minimize the effects of the epidemic	Contribute to productive efforts for change
		Administrative
		Sharing of information across institutions/systems
		Peer mentorship for clinical, administrative, and academic
		duties related to COVID-19
		Creation of volunteering, innovation, and service
		opportunities to support response efforts (e.g., creation of
		new devices/tools, clinical strategies)

From Wong [80], with permission Elsevier

Palliative Care

Emergency physicians encounter patients with unmet palliative care needs during almost all clinical shifts [81]. Given that the majority of hospital admissions originate in the ED, critical and timely decision-making by emergency physicians often determines the subsequent intensity and trajectory of treatment for life-limiting illnesses, including intubation and intensive care utilization. During the COVID-19 pandemic, questions of medical futility and appropriate goals of care have become more prominent and efforts to address unresolved palliative care issues have received increasing attention.

Early in the pandemic, the University of Washington implemented an institution-wide plan to address COVID-19-related palliative care needs [82]. Focused on the ED, the intensive care unit, and acute care services, it provided strategies to identify goals of care, manage distressing symptoms, and support family members. In the ED, the palliative care service conducted daily huddles with staff, provided consultations for patients with poor prognosis and at risk for intubation or CPR, supported implementation of DNR orders when appropriate, and embedded a palliative care specialist to screen patients. The palliative care service also provided telephonic coaching and support 24 hours/day, 7 days/week.

In responding to the early surge of COVID-19, New York-Presbyterian Columbia University Irving Medical Center developed a psychiatry-palliative care liaison team (including trainees) to provide ED palliative care services [83]. During the peak of the epidemic, this service provided a 24-hour coverage for palliative care needs, allowing emergency providers to better meet the overwhelming demand for medical care and establishing a model to provide future surge capacity. For 110 ED patients seen by the palliative care team, the median age was 81.5 (range 46–101), twothirds were community dwelling, and few presented with advance directives or Medical Orders for Life-Sustaining Treatment (MOLST) [84]. Ninety-one of 110 patients (83%) were considered "full code" on arrival to the ED, falling to 20 of 100 (18%) after the initial ED palliative care consultation.

In a qualitative study, emergency medicine researchers interviewed representatives from 52 hospitals in the USA to identify ED-palliative care innovations driven by the pandemic [85].

Table 70.3 summarizes these innovations in systems, staffing, and technology. Respondents reported that such innovation was welcomed by clinicians and positively impacted patient care trajectories.

Conclusions

In this brief review, we have discussed a number of challenges posed by the COVID-19 pandemic and its impact on the evolving subdiscipline of oncologic emergency medicine. Beyond direct negative outcomes due to infection among our patients, we are concerned that degradation of our screening and prevention efforts, in addition to pandemicrelated barriers to appropriate cancer treatment and research, will have long-lasting effects well into the post-pandemic era. At the same time, the pandemic has exposed faults within the healthcare system and society that, when recognized and confronted, hold the promise of a better future.

In the USA specifically, the politicization of science, with its long and dark history, has become increasingly overt. This anti-science effort must be resisted and counteractions taken against it by physicians, researchers, and cancer patients. The stark evidence of health disparities associated with COVID-19 should give rise to further efforts to reduce and eliminate such inequities. Given resource constraints posed by the virus, our health systems have an opportunity to reexamine the consequences of overly aggressive cancer therapies, as well as the inferior provision of palliative care. A necessary redesign of our cancer care system in response to COVID-19 has many lessons for the larger healthcare system and society at large.

Table 70.3 Summary of innovations in ED palliative care

	Example of	
Type of innovation	innovation	Innovation detail
Model of care delivery	Embedded PC clinician in the ED	PC clinician seated in the ED dedicated only to ED consults
	Strengthened ED presence	Achieved through daily rounding, EMR chat function
	Mobile PC consult service	Dedicated service focused on ED and ICU needs
Staffing	PC attendings with extenders	Residents with focused GOC or ACP training
	PC attending with PC fellows	Triage cases based on complexity to appropriate clinician
	PC extender with psychosocial partner	Pair volunteer non-PC physician with social worker or child life specialist who perform all consults together
Technology- enhanced PC-ED	Off-site tele-PC	Centralized team of either RNs or PC physicians for all hospitals in a health system
	Blended on-site tele-PC	Triage patients based on their capacity to engage to either in person or tele-PC
Primary PC training and education	Trainings and tools	COVID-specific conversation training; collated resources (with apps, Google Docs, provided laminated cards)
Case identification and task	Proactive case identification	Remotely screen ED track board, daily rounding
stratification	Formal triggers (for primary PC or specialty consult)	Automated or manual – encompassing age, marker of underlying illness, marker of acute illness
	Focused, abbreviated consults	Task-oriented consults focused on specific patient needs
	Nursing-initiated consults	Consults to PC triggered by nursing staff using clear trigger criteria

From Aaronson [85] with permission Elsevier

References

- World Health Organization. WHO coronavirus disease (COVID-19) dashboard. Updated 11 Nov 2020. https://covid19.who.int/. Accessed 17 Nov 2020.
- The New York Times. Covid in the U.S.: latest map and case count. https://www.nytimes.com/interactive/2020/us/coronavirusus-cases.html?name=styln-coronavirus®ion=TOP_ BANNER&block=storyline_menu_recirc&action=click&p gtype=Article&impression_id=655afd10-2926-11eb-9637cd5e2c4edeed&variant=1_Show. Accessed 17 Nov 2020.
- The Institute for Health Metrics and Evaluation. COVID-19 projections. https://covid19.healthdata.org/united-states-of-america. Accessed 17 Nov 2020.

- Muennig PA, Reynolds M, Fink DS, Zafari Z, Geronimus AT. America's declining well-being, health, and life expectancy: not just a white problem. Am J Public Health. 2018;108(12):1626–31.
- Selden TM, Berdahl TA. COVID-19 and racial/ethnic disparities in health risk, employment, and household composition. Health Aff (Millwood). 2020;39(9):1624–32.
- Derosa L, Melenotte C, Griscelli F, et al. The immuno-oncological challenge of COVID-19. Nat Cancer. 2020;1:946–64.
- Giesen N, Sprute R, Rüthrich M, Khodamoradi Y, Mellinghoff SC, Beutel G, et al. Evidence-based management of COVID-19 in cancer patients: Guideline by the Infectious Diseases Working Party (AGIHO) of the German Society for Haematology and Medical Oncology (DGHO). Eur J Cancer. 2020;140:86–104.
- Tian J, Yuan X, Xiao J, Zhong Q, Yang C, Liu B, et al. Clinical characteristics and risk factors associated with COVID-19 disease severity in patients with cancer in Wuhan, China: a multicentre, retrospective, cohort study. Lancet Oncol. 2020;21(7):893–903.
- Kuderer NM, Choueiri TK, Shah DP, Shyr Y, Rubinstein SM, Rivera DR, COVID-19 and Cancer Consortium, et al. Clinical impact of COVID-19 on patients with cancer (CCC19): a cohort study. Lancet. 2020;395(10241):1907–18. Erratum in: Lancet. 2020;396(10253):758.
- Mehta V, Goel S, Kabarriti R, Cole D, Goldfinger M, Acuna-Villaorduna A, et al. Case fatality rate of cancer patients with COVID-19 in a New York hospital system. Cancer Discov. 2020;10(7):935–41.
- Aries JA, Davies JK, Auer RL, Hallam SL, Montoto S, Smith M, et al. Clinical outcome of coronavirus disease 2019 in haematooncology patients. Br J Haematol. 2020;190(2):e64–7.
- Yang K, Sheng Y, Huang C, Jin Y, Xiong N, Jiang K, et al. Clinical characteristics, outcomes, and risk factors for mortality in patients with cancer and COVID-19 in Hubei, China: a multicentre, retrospective, cohort study. Lancet Oncol. 2020;21(7):904–13.
- Assaad S, Avrillon V, Fournier ML, Mastroianni B, Russias B, Swalduz A, et al. High mortality rate in cancer patients with symptoms of COVID-19 with or without detectable SARS-COV-2 on RT-PCR. Eur J Cancer. 2020;135:251–9.
- Garassino MC, Whisenant JG, Huang LC, Trama A, Torri V, Agustoni F, et al. COVID-19 in patients with thoracic malignancies (TERAVOLT): first results of an international, registry based, cohort study. Lancet Oncol. 2020;21(7):914–22.
- Meng Y, Lu W, Guo E, Liu J, Yang B, Wu P, et al. Cancer history is an independent risk factor for mortality in hospitalized COVID-19 patients: a propensity score-matched analysis. J Hematol Oncol. 2020;13(1):75.
- He W, Chen L, Chen L, Yuan G, Fang Y, Chen W, et al. COVID-19 in persons with haematological cancers. Leukemia. 2020;34(6):1637–45.
- Martin-Moro F, Marquet J, Piris M, Michael BM, Sa'ez AJ, Corona M, et al. Survival study of hospitalised patients with concurrent COVID-19 and haematological malignancies. Br J Haematol. 2020;190(1):e16–20.
- Fattizzo B, Giannotta JA, Sciumè M, Cattaneo D, Bucelli C, Fracchiolla NS, et al. Reply to "COVID-19 in persons with haematological cancers": a focus on myeloid neoplasms and risk factors for mortality. Leukemia. 2020;34(7):1957–60.
- Venkatesulu BP, Chandrasekar VT, Girdhar P, Advani P, Sharma A, Elumalai T, et al. A systematic review and meta-analysis of cancer patients affected by a novel coronavirus. medRxiv. [Preprint]. 2020; https://doi.org/10.1101/2020.05.27.20115303.
- Williamson EJ, Walker AJ, Bhaskaran K, Bacon S, Bates C, Morton CE, et al. Factors associated with COVID-19-related death using OpenSAFELY. Nature. 2020;584(7821):430–6.
- 21. Lee LYW, Cazier JB, Starkey T, Briggs SEW, Arnold R, Bisht V, UK Coronavirus Cancer Monitoring Project Team, et al. COVID-19 prevalence and mortality in patients with cancer and the effect of

primary tumour subtype and patient demographics: a prospective cohort study. Lancet Oncol. 2020;21(10):1309–16.

- Dai M, Liu D, Liu M, Zhou F, Li G, Chen Z, et al. Patients with cancer appear more vulnerable to SARS-CoV-2: a multicenter study during the COVID-19 outbreak. Cancer Discov. 2020;10(6):783–91.
- 23. Zhang L, Zhu F, Xie L, Wang C, Wang J, Chen R, et al. Clinical characteristics of COVID-19-infected cancer patients: a retrospective case study in three hospitals within Wuhan, China. Ann Oncol. 2020;31(7):894–901.
- Tang LV, Hu Y. Poor clinical outcomes for patients with cancer during the COVID-19 pandemic. Lancet Oncol. 2020;21(7):862–4.
- Yang F, Shi S, Zhu J, Shi J, Dai K, Chen X. Clinical characteristics and outcomes of cancer patients with COVID-19. J Med Virol. 2020;92(10):2067–73. https://doi.org/10.1002/jmv.25972. Epub ahead of print.
- 26. Yarza R, Bover M, Paredes D, Lopez-Lopez F, Jara-Casas D, Castelo-Loureiro A, et al. SARS-CoV-2 infection in cancer patients undergoing active treatment: analysis of clinical features and predictive factors for severe respiratory failure and death. Eur J Cancer. 2020;135:242–50.
- 27. Nawar T, Morjaria S, Kaltsas A, Patel D, Perez-Johnston R, Daniyan AF, et al. Granulocyte-colony stimulating factor in COVID-19: is it stimulating more than just the bone marrow? Am J Hematol. 2020;95(8):E210–3.
- Liang W, Guan W, Chen R, Wang W, Li J, Xu K, et al. Cancer patients in SARS-CoV-2 infection: a nationwide analysis in China. Lancet Oncol. 2020;21(3):335–7.
- 29. Nichetti F, Bini M, Ambrosini M, Ottini A, Rametta A, Leporati R, et al. COVID-19 risk for patients undergoing anticancer treatment at the outpatient clinic of the National Cancer Institute of Milan: the COVINT study. ESMO Open. 2020;5(Suppl 3):e000883.
- 30. Rüthrich MM, Giessen-Jung C, Borgmann S, Classen AY, Dolff S, Grüner B, LEOSS Study Group, et al. COVID-19 in cancer patients: clinical characteristics and outcome-an analysis of the LEOSS registry. Ann Hematol. 2021;100(2):383–93. https://doi.org/10.1007/s00277-020-04328-4. Epub ahead of print.
- Meehan MC. The butterfly caste: a social history of pellagra in the south. JAMA. 1973;223(1):84.
- Price-Haywood EG, Burton J, Fort D, Seoane L. Hospitalization and mortality among black patients and white patients with Covid-19. N Engl J Med. 2020;382(26):2534–43.
- 33. Azar KMJ, Shen Z, Romanelli RJ, Lockhart SH, Smits K, Robinson S, Brown S, Pressman AR. Disparities in outcomes among covid-19 patients in a large health care system in California. Health Aff (Millwood). 2020;39(7):1253–62.
- 34. Rivera DR, Peters S, Panagiotou OA, Shah DP, Kuderer NM, Hsu CY, COVID-19 and Cancer Consortium, et al. Utilization of COVID-19 treatments and clinical outcomes among patients with cancer: a COVID-19 and Cancer Consortium (CCC19) Cohort Study. Cancer Discov. 2020;10(10):1514–27.
- Tirupathi R, Muradova V, Shekhar R, Salim SA, Al-Tawfiq JA, Palabindala V. COVID-19 disparity among racial and ethnic minorities in the US: a cross sectional analysis. Travel Med Infect Dis. 2020;38:101904. https://doi.org/10.1016/j.tmaid.2020.101904. Epub ahead of print.
- Rodriguez RM. Tackling another COVID-19 pandemic disparity: distance from major academic medical centers encumbers emergency and critical care physician surge capacity. Acad Emerg Med. 2020;27:1212–4. https://doi.org/10.1111/acem.14123. Epub ahead of print.
- 37. Yancy CW. COVID-19 and African Americans. JAMA. 2020;323(19):1891–2.
- Owen WF, Carmona R, Pomeroy C. Failing another national stress test on health disparities. JAMA. 2020;323(19):1905–6.
- Leiker B, Wise K. COVID 19 case study in emergency medicine preparedness and response: from personal protective equip-

ment to delivery of care. Dis Mon. 2020;66(9):101060. https://doi. org/10.1016/j.disamonth.2020.101060. Epub ahead of print.

- 40. Millard WB. Surgeproofing the hot zone: preparing for a second wave of COVID-19. Ann Emerg Med. 2020;76(4):A19–22.
- 41. Anesthesia Patient Safety Foundation. Perioperative considerations for the 2019 novel coronavirus (COVID-19). Anesthesia Patient Safety Foundation. Published 12 Feb 2020. https://www. apsf.org/news-updates/perioperative-considerations-for-the-2019novelcoronavirus-covid-19/. Accessed 16 Nov 2020.
- 42. Ghelichkhani P, Esmaeili M. Prone position in management of covid-19 patients; a commentary. Arch Acad Emerg Med. 2020;8(1):e48.
- Widysanto A, Wahyuni TD, Simanjuntak LH, et al. Happy hypoxia in critical COVID-19 patient: a case report in Tangerang, Indonesia. Physiol Rep. 2020;8(20):e14619.
- 44. Brown CA 3rd, Bair AE, Pallin DJ, Walls RM, NEAR III Investigators. Techniques, success, and adverse events of emergency department adult intubations. Ann Emerg Med. 2015;65(4):363–70. e1.
- 45. Weisberg E, Parent A, Yang PL, Sattler M, Liu Q, Liu Q, et al. Repurposing of kinase inhibitors for treatment of COVID-19. Pharm Res. 2020;37(9):167.
- Broom A, Kenny K, Page A, Cort N, Lipp ES, Tan AC, et al. The paradoxical effects of COVID-19 on cancer care: current context and potential lasting impacts. Clin Cancer Res. 2020;26(22):5809–13.
- 47. Percival MM, Lynch RC, Halpern AB, Shadman M, Cassaday RD, Ujjani C, et al. Considerations for managing patients with hematologic malignancy during the COVID-19 pandemic: the Seattle strategy. JCO Oncol Pract. 2020;16(9):571–8.
- van de Haar J, Hoes LR, Coles CE, Seamon K, Fröhling S, Jäger D, et al. Caring for patients with cancer in the COVID-19 era. Nat Med. 2020;26(5):665–71. https://doi.org/10.1038/s41591-020-0874-8. Erratum: Nat Med. 2020;26(7):1146.
- 49. Minder R. Spain's other COVID casualties: undetected cancer cases. New York Times. 16 Nov 2020. https://www.nytimes.com/2020/11/16/world/europe/spain-coronavirus-cancer.html. Accessed 16 Nov 2020.
- Cancer Action Network, American Cancer Society. COVID-19 pandemic impact on cancer patients and survivors: survey findings summary. March–April, 2020. https://www.fightcancer.org/ sites/default/files/National%20Documents/Survivor%20Views. COVID19%20Polling%20Memo.Final_.pdf. Accessed 16 Nov 2020.
- London JW, Fazio-Eynullayeva E, Palchuk MB, Sankey P, McNair C. Effects of the COVID-19 pandemic on cancer-related patient encounters. JCO Clin Cancer Inform. 2020;4:657–65.
- 52. Norbash AM, Moore AV Jr, Recht MP, Brink JA, Hess CP, Won JJ, et al. Early-stage radiology volume effects and considerations with the coronavirus disease 2019 (COVID-19) pandemic: adaptations, risks, and lessons learned. J Am Coll Radiol. 2020;17(9):1086–95.
- Hollander JE, Carr BG. Virtually perfect? Telemedicine for Covid-19. N Engl J Med. 2020;382(18):1679–81.
- Royce TJ, Sanoff HK, Rewari A. Telemedicine for cancer care in the time of COVID-19. JAMA Oncol. 2020;6(11):1698–9. https:// doi.org/10.1001/jamaoncol.2020.2684. Epub ahead of print.
- Singer DS. NCI's work to advance cancer research while responding to the COVID-19 pandemic. Cancer Cell. 2020;37(6):746–8.
- Bakouny Z, Hawley JE, Choueiri TK, Peters S, Rini BI, Warner JL, Painter CA. COVID-19 and cancer: current challenges and perspectives. Cancer Cell. 2020;38(5):629–46.
- Richards M, Anderson M, Carter P, Ebert BL, Mossialos E. The impact of the COVID-19 pandemic on cancer care. Nat Cancer. 2020;1:565–7. https://doi.org/10.1038/s43018-020-0074-y. Epub ahead of print.
- Centers for Medicare & Medicaid Services. Non-emergent, elective medical services, and treatment recommendations. 7 Mar 2020.

https://www.cms.gov/files/document/cms-non-emergent-electivemedical-recommendations.pdf. Accessed 17 Nov 2020.

- 59. Alla F, Berlin I, Nguyen-Thanh V, Guignard R, Pasquereau A, Quelet S, et al. Tobacco and COVID-19: a crisis within a crisis? Can J Public Health. 2020;111(6):995–9. https://doi.org/10.17269/ s41997-020-00427-x. Epub ahead of print.
- 60. Bommele J, Hopman P, Walters BH, Geboers C, Croes E, Fong GT, et al. The double-edged relationship between COVID-19 stress and smoking: implications for smoking cessation. Tob Induc Dis. 2020;18:63.
- Masroor S. Collateral damage of COVID-19 pandemic: delayed medical care. J Card Surg. 2020;35(6):1345–7.
- 62. Farsalinos K, Barbouni A, Niaura R. Systematic review of the prevalence of current smoking among hospitalized COVID-19 patients in China: could nicotine be a therapeutic option? Intern Emerg Med. 2020;15(5):845–52.
- 63. Petrilli CM, Jones SA, Yang J, Rajagopalan H, O'Donnell L, Chernyak Y, et al. Factors associated with hospital admission and critical illness among 5279 people with coronavirus disease 2019 in New York City: prospective cohort study. BMJ. 2020;369:m1966.
- 64. Murray G, Roche D, Ridge A, Hackett C, Tobin AM. Response to "Reduction in skin cancer diagnosis, and overall cancer referrals, during the COVID-19 pandemic" (letter). Br J Dermatol. 2020; https://doi.org/10.1111/bjd.19667. Epub ahead of print.
- 65. Earnshaw CH, Hunter HJA, McMullen E, Griffiths CEM, Warren RB. Reduction in skin cancer diagnosis, and overall cancer referrals, during the COVID-19 pandemic. Br J Dermatol. 2020;183(4):792–4.
- 66. Baugh JJ, White BA, McEvoy D, Yun BJ, Brown DFM, Raja AS, Dutta S. The cases not seen: patterns of emergency department visits and procedures in the era of COVID-19. Am J Emerg Med. 2020; https://doi.org/10.1016/j.ajem.2020.10.081. Epub ahead of print.
- 67. Jeffery MM, D'Onofrio G, Paek H, Platts-Mills TF, Soares WE 3rd, Hoppe JA, et al. Trends in emergency department visits and hospital admissions in health care systems in 5 states in the first months of the COVID-19 pandemic in the US. JAMA Intern Med. 2020;180(10):1328–33.
- Westgard BC, Morgan MW, Vazquez-Benitez G, Erickson LO, Zwank MD. An analysis of changes in emergency department visits after a state declaration during the time of COVID-19. Ann Emerg Med. 2020;76(5):595–601.
- Hartnett KP, Kite-Powell A, DeVies J, Coletta MA, Boehmer TK, Adjemian J, Gundlapalli AV, National Syndromic Surveillance Program Community of Practice. Impact of the COVID-19 pandemic on emergency department visits – United States, January 1, 2019-May 30, 2020. MMWR Morb Mortal Wkly Rep. 2020;69(23):699–704.
- Kliff S. Two friends in Texas were tested for coronavirus. One bill was \$199. The other? \$6,408. The New York Times. 29 June 2020. Updated 15 July 2020. https://www.nytimes.com/2020/06/29/ upshot/coronavirus-tests-unpredictable-prices.html. Accessed 17 Nov 2020.
- 71. Kliff S. How to avoid a surprise bill for your coronavirus test. The New York Times. 13 Nov 2020. https://www.nytimes. com/2020/11/13/upshot/coronavirus-surprise-bills-guide.html. Accessed 17 Nov 2020.

- Pines JM. COVID-19, Medicare for all, and the uncertain future of emergency medicine. Ann Emerg Med. 2020;76(4):459–61.
- The Physicians Foundation. 2020 survey of America's physicians: COVID-19 impact edition. Aug 2020. https://physiciansfoundation. org/wp-content/uploads/2020/08/20-1278-Merritt-Hawkins-2020-Physicians-Foundation-Survey.6.pdf. Accessed 16 Nov 2020.
- 74. Gafford JA, Gurley-Calvez T, Krebill H, Lai SM, Christiadi, Doolittle GC. Expanding local cancer clinical trial options: analysis of the economic impact of the Midwest Cancer Alliance in Kansas. Acad Med. 2017;92(9):1274–9.
- Stehman CR, Testo Z, Gershaw RS, Kellogg AR. Burnout, drop out, suicide: physician loss in emergency medicine, Part I. West J Emerg Med. 2019;20(3):485–94.
- 76. Lai J, Ma S, Wang Y, Cai Z, Hu J, Wei N, et al. Factors associated with mental health outcomes among health care workers exposed to coronavirus disease 2019. JAMA Netw Open. 2020;3(3):e203976.
- 77. Jeong JM, Radeos MS, Shee B, Kindschuh M, Sasson C, Hernandez C, et al. COVID-19 seroconversion in emergency professionals at an urban academic emergency department in New York city. Ann Emerg Med. 2020;76:815–6. https://doi.org/10.1016/j.annemerg-med.2020.06.038. Epub ahead of print.
- Dean W, Jacobs B, Manfredi RA. Moral injury: the invisible epidemic in COVID health care workers. Ann Emerg Med. 2020;76(4):385–6.
- 79. Rodriguez RM, Medak AJ, Baumann BM, Lim S, Chinnock B, Frazier R, Cooper RJ. Academic emergency medicine physicians' anxiety levels, stressors, and potential stress mitigation measures during the acceleration phase of the COVID-19 pandemic. Acad Emerg Med. 2020;27(8):700–7. https://doi.org/10.1111/ acem.14065. Epub ahead of print.
- Wong AH, Pacella-LaBarbara ML, Ray JM, Ranney ML, Chang BP. Healing the healer: protecting emergency health care workers' mental health during COVID-19. Ann Emerg Med. 2020;76(4):379–84.
- 81. Todd KH. Practically speaking: emergency medicine and the palliative care movement. Emerg Med Australas. 2012;24(1):4–6.
- 82. Fausto J, Hirano L, Lam D, Mehta A, Mills B, Owens D, et al. Creating a palliative care inpatient response plan for COVID-19-the UW medicine experience. J Pain Symptom Manag. 2020;60(1):e21–6.
- 83. Shalev D, Nakagawa S, Stroeh OM, Arbuckle MR, Rendleman R, Blinderman CD, Shapiro PA. The creation of a psychiatry-palliative care liaison team: using psychiatrists to extend palliative care delivery and access during the COVID-19 crisis. J Pain Symptom Manag. 2020;60(3):e12–6.
- Lee J, Abrukin L, Flores S, Gavin N, Romney ML, Blinderman CD, Nakagawa S. Early intervention of palliative care in the emergency department during the COVID-19 pandemic. JAMA Intern Med. 2020;180(9):1252–4.
- Aaronson EL, Daubman BR, Petrillo L, Bowman J, Ouchi K, Gipps A, et al. Emerging palliative care innovations in the ED: a qualitative analysis of programmatic elements during the COVID-19 pandemic. J Pain Symptom Manage. 2020:S0885-3924(20)30864-2. https://doi.org/10.1016/j.jpainsymman.2020.10.035. Epub ahead of print.



The Physician and Cancer: In Their Own Words

71

Marshall T. Morgan, Patrick J. Crocker, Burton F. Dickey, Sherry-Ann Brown, and Knox H. Todd

Introduction

The word, "cancer" is, in itself, a powerful term that should never be underestimated. It carries a certain shock value – a sense of doom and gloom that seems qualitatively different from other diagnoses. This is as true for physicians as it is for laypersons. In this collection of prose and poetry, four physicians tell us stories of their cancer experiences, with the goal of communicating what matters. We hope these illness narratives help our readers care for themselves and for others in a more human, and humane, manner.

Editor's Note

Marshall Morgan was my mentor and friend. He headed the UCLA emergency medicine program where I trained, and throughout our careers we met frequently to share stories of our lives, usually over a meal and always over a glass of bourbon. By example, Marshall taught me the importance of empathy. Professionally, one of the things I most admired

M. T. Morgan (Deceased)

P. J. Crocker Department of Emergency Medicine, Dell Children's Medical Center, Austin, TX, USA

Department of Emergency Medicine, Brackenridge Hospital, Austin, TX, USA

B. F. Dickey

Department of Pulmonary Medicine, The University of Texas MD Anderson Cancer Center, Houston, TX, USA

S.-A. Brown

Cardio-Oncology Program, Division of Cardiovascular Medicine, Department of Medicine, Medical College of Wisconsin, Milwaukee, WI, USA

K. H. Todd (⊠) (Retired) Department of Emergency Medicine, MD Anderson Cancer Center, Houston, TX, USA in Marshall was that although he was generally the smartest person in the room, you didn't realize it until later. The day after he felt the first symptoms of cancer, Marshall took time to write the words below. These were included in his memorial service a short time later.

The Difference a Day Makes

Marshall T. Morgan, MD (1941-2015)

Yesterday morning thinking about my future I made the reasonable estimate that I would live in a functioning active state into my mid-to-late 80s (I am 73). I considered: When should I retire (not now: my wife is still working and I like my job)? On the other hand, I spend too little time with my children and grandchildren, so maybe sooner than later.

Last night, I developed an unusual RUQ pain with a strong pleuritic component and no tenderness. I consulted my physician and we decided I should go to the ED. The ED workup, which would have taken a week as an outpatient but was done in 4 hours, revealed a large number of liver lesions and a mass hugging the lesser curvature of the stomach.

Today, unlike yesterday, I am contemplating a much shorter future for myself (disappointingly I have young grandchildren). I will undergo EGD tomorrow in an attempt to get a tissue diagnosis and soon will face the question of chemotherapy, whether to undergo chemotherapy.

As an emergency physician I have been the messenger to others on the day that made the difference. Some, like the 60-year-old gentleman with a large palpable right upper quadrant mass, were not surprised at all, but resigned. Others whose imaging was done in the expectation of finding a gallstone or kidney stone were shocked as I was. Interestingly, when there were tears, it was family members, not the patient, who shed them.

I have shed no tears. Most of my thoughts have been around changing the expectation of my remaining life span and hoping that I will be spared severe pain and disability

Division of Emergency Medicine, Department of Medicine, UCLA Medical Center, Los Angeles, CA, USA

for some time, so I can enjoy my family, my friends, and this wonderful planet to the greatest extent possible before I leave them behind.

Jean

Marie

M.T.M. February 5, 2015 Los Angeles (Reprinted with permission from Campbell-Morgan)

Editor's Note

Pat Crocker is a true pioneer of emergency medicine in Central Texas, having served as Chief of Emergency Medicine at both Austin's Brackenridge Hospital and later, Dell Children's Medical Center. He now lives in Kerrville, Texas, and recently published two books of stories from his rich career [1, 2]. On reviewing his second book: *More Letters from the Pit: Stories of a Physician's Odyssey in Emergency Medicine*, I wrote:

... Pat Crocker captures the stories of our lives as emergency physicians. These accounts of the highs and lows in caring for patients over his remarkable career are delivered in often stark, and occasionally grim, terms. Physician readers will recognize these stories immediately, if not as their own, then as those that might have been. For readers with a non-medical background, reading these pages opens a rich window into the human theater that is emergency medicine. For our families and loved ones, reading this book will help you understand why we might be less talkative than usual when returning from a long clinical shift. My hope is that after reading this book, conversations about our work become easier. At times tragic, at times humorous, More Letters from the Pit conveys a profound understanding of the gratitude we owe our patients and a sense of wonder at the mysteries of chance and providence that bring us together in the crucible that is the modern emergency department [2].

Thomas J.

Patrick J. Crocker, DO

I have found that some of the great saves are not those accompanied by that adrenalin rush of the ER we in the field seem to crave. No blood and guts. Instead they're ones fighting foes that threaten to kill your patient from the inside out. Cases that require thought, planning, an engaged patient, and a whole team of multidisciplinary players. These kinds of saves can roll out over months and the ending uncertain for years. But they are no less gratifying than the others.

I first met Thomas as Marcia and I prepared to build our new house. We had decided on polished concrete floors, and our builder who referred us to him thought he was the best in the business. And so we drove out to meet him, see his shop, and let him show us some samples of his work.

The shop was located out on a country road. Rather an informal place surrounded by beautiful old oak trees and, of course, displays of his handiwork. We walked around a bit and saw no one, so went inside to the office area. No one there either. "Hello," I called out. And then a little shuffling from a back room and he emerged from behind a curtain.

He smiles and says hello. A firm handshake. He's about my height, fit, slender, and rangy. I can tell by looking at his sloping shoulders and biceps he works for a living. And obviously doesn't think concrete work is enough exercise so he hits the weights regularly too. He seems a bit rough around the edges. I suspect he likes to party, ride motorcycles, and an occasional bar scuffle may not be beyond him. I like him immediately.

He's anxious to show us his craft and we walk around his shop seeing what he can do. He is a master. This guy turns concrete into something almost indistinguishable from marble. Creates beautiful concrete countertops, tabletops, and sinks. It only takes us a few minutes to agree this is our man and we tell him we'll let the builder know to call him.

The work on the house goes forth and I get to know Thomas better. He proves to be a very likeable sort. Smart, hardworking, and honest. A man of his word. After dealing with a few issues that cost him some profit, I thanked him for being one of the few honest folks in the construction business that I've ever met. We became friends over the months of building.

One day we're talking about medicine, my experiences at the trauma center, and he asks if I could be his doctor. He's a hardworking guy but without medical insurance so I said sure, but I can't see him at the ER. I told him I would do a physical, get some labs, and get him on the right track. He hadn't seen a doctor in a decade and it was past time. He agreed and I told him to call when he was ready. The physical would be in my office at the new house.

Months pass and I don't hear from him. I supposed the same distaste for doctors and health care that kept him away for a decade had not yet been overcome. And then a day later, the phone rings. He wants to know if I could have a look at his elbow following an injury. And I of course said come on over. And he does.

It's a sad conundrum for physicians to help patients like this. In my case, my malpractice insurance only covers me for cases in the hospital. I'll be going bare to help him out but he deserves it. I also trust him to believe that no matter what the result may turn out to be, I did my best. Nevertheless, we will have the conversation that my no-cost assistance comes with the agreement he forgo the right to sue me. I hate to bring this up with someone I now consider a friend but I also have to protect my family. He will understand.

Thomas shows up a few hours later sporting his usual smile and laugh. He's still ten feet away but I can already see his elbow is a mess. He tells what he can remember of the incident, which isn't much. He was at the Burning Man festival and then woke up in jail. We both laugh. He says all he really knows is when he was released he found a police citation in his pocket. I laugh to myself at this one. Good luck, Nevada police, you'll probably never be seeing Thomas again. This is *pure* Thomas!

After I examine him I conclude he has probably torn part of his bicep insertion and wrenched the joint pretty badly. I get him set up with an orthopedist I went to medical school with. As he is captive in my office I ask him about that physical I was to perform for him. I say, "Dude, you're 50 years old and a smoker. We need to do this for you. Don't be a stupid shit and put this off any longer." We are obviously frank with each other. I also point out he won't be working for a while and it's the perfect time to go get that lab work done. And that I'll want a chest X-ray. I'm not going to take responsibility for a smoker's health at his age without one. He laughs his laugh and agrees, and the next day follows up as he said he would. Thomas is a man of his word.

Two days later all the lab work has returned and is normal. I didn't expect to find much but when a patient hasn't seen a physician in so long you never know. I also get unwelcome news.

One of the radiologists saw my name on the chest X-ray order and decided to call me directly. He spills the news that it shows a mass in his right lung and it looks like whatever it is has spread to the lymph nodes in the right side of the chest. This means most likely cancer and fairly advanced. This is horrible. I worry it's a death sentence for him.

I give him a call, trying not to alarm him but tell him what I have found. Bad news is something I dread giving patients. But we must deal with the facts. No exceptions. I tell him to head back to the radiologist for an enhanced CT scan of his chest. We'll try to nail this down as best we can as soon as possible. Thomas takes the news straight up. He knows I'm worried. I order a CT scan of his chest and he returns to the radiologist.

Late that afternoon the radiologist is on the phone again. He's confirming on the CT scan this appears to be lung cancer, spread to the lymph nodes on the right side. Even worse it appears to have involved the aorta. This may make the cancer inoperable, I fear.

Before I tell Thomas the bad news I call one of the oncologists I've known for years. I consider him one of the best and would trust him with my family members. We discuss the case and he isn't too hopeful. Statistically, he suspects a small cell carcinoma, too advanced for surgery, and a very low survival rate. I feel crushed for Thomas. He does add, however, that given his young age, he thinks we should go for broke on this one, try for a save. Not palliation where we might just try to give him some more time on earth but go for a straight-up cure. He asks me to send him over to the office.

A PET scan is ordered and the oncologist and I speak again. Yes, definitely a cancer. Thomas will now need a biopsy of the lesion to nail down the type and characteristics of the cancer. He again takes the news in a levelheaded fashion and then asks the question I dread. "How long do you think I've got Doc?" I hedge a bit, let him know it's not good news, but the oncologist wants to go for a cure. And Thomas is now determined to get that cure. I add one other thing, "You must quit smoking NOW. The oncologist says there's no hope if you keep smoking." He says he will quit today.

The rest of the story unfolds over the following weeks, pretty much as predicted. The biopsy confirms small cell carcinoma and the prognosis is not good. The big decision is what to do for Thomas. Surgical removal is not possible.

A patient's active involvement in their healthcare is something I treasure. I believe an active participating patient leads to better outcomes. Thomas and the oncologist talk over treatment options, all of which are aggressive. Thomas and I review everything. Before starting treatment Thomas wants to consult with some other specialists. I agree, but say, "Let's not waste a lot of time." We'll be thorough but I want to get things started. Thomas is making his survival his only project, tracking down every available approach. He even includes an herbalist. Checking into supplements that might increase his chance of survival or at least keep him healthier through the chemo process.

A couple of weeks pass as other consultants see him. He finds another very aggressive oncologist who wants to press for a cure also. The two oncologists disagree a bit on therapy, as they don't want to kill him during the process, but finally agree to take the chance. And so does Thomas. The goal is a cure if possible. A combination of chemotherapy and radiation that will begin immediately.

Through 5 rigorous months of therapy, including prophylactic brain radiation to kill any cancer cells hiding there, Thomas and I talk from time to time. I give him my encouragement. I learn he's still hanging in there, still determined despite feeling poorly. We talk about the predictable baldness, all the economic stress of this kind of major therapy, and the supplements. Multiple PET scans and follow-ups occur. The tumor shrinks but remains visible on the scans. After a year, however, it hasn't progressed and we start to consider that maybe this is residual dead tumor and scarring. Another year goes by without any change and happily the tumor ultimately is considered dead. I'm elated for Thomas.

So this case had none of the adrenalin we ER doctors seem to crave. It was a very satisfying result nonetheless. The excitement of blood and guts substituted for a thorough approach by a great team of doctors and a very determined patient.

I am again convinced that fate plays a role in our lives. Also that thoroughness and an unflinching determination to cover all your bases from the start, every time, save patient's lives. As usual, all of the what if's churn through my mind. What if I had let Thomas's initial reluctance to skip the chest X-ray rule? Just entered a note, "Patient refuses chest X-ray at this time." Covered my ass with a simple note and left his hanging in the breeze. I just can't approach patients like that and glad I didn't.

I recall a patient I saw years before who presented with a complaint of a shoulder rash. On examination I found a very large melanoma lesion. I explained what was wrong and what we needed to do. She refused, saying she would treat the lesion with over-the-counter peroxide. We talked for about 15 minutes, me attempting to convince her that peroxide treatment was not only ill-advised but would result in her death. She was adamant. Unbelievable.

And I entered that CYA, cover your ass, note in the chart, "Patient refuses referral for an apparent advanced melanoma despite counseling" and had her signed out against medical advice. A sad addition to her medical record. My ass was covered, but hers wasn't. I consider such encounters personal failures every time. I have failed to engage my patient in a fashion that led them to the appropriate care.

Thomas remains happy, healthy, and tumor-free. We talk once in a while. Someday when we're both motivated he'll give me some concrete finishing lessons. We'll build a tabletop together.

And though the danger of getting too close to your patient and their tragedy and crossing that line between empathy and self-preservation is one I wrestle with almost daily, sometimes you have to cross it for a friend.

Editor's Note

As a fellow chair at MD Anderson, I will always be indebted to Burton Dickey for his advice on managing my department within a byzantine modern cancer center and for his everquick wit and humor that sustained me through interminable administrative meetings. When I asked Burton to write about his experience with multiple myeloma (from which he is in remission and doing well), he told me: "Don't expect anything sentimental or melodramatic; that's not me. Just the facts." What follows is his advice on what to do (and what not to do) for physicians facing a cancer diagnosis.

Advice for a Physician (or Anyone Else) with Newly Diagnosed Cancer

Burton F. Dickey, MD

I was diagnosed with multiple myeloma 8 years ago and today am fortunate to be healthy and in complete remission. Quite a few physicians, including my friend Knox Todd, have asked what I learned. My advice comes down to three simple suggestions. First, read about your disease – a lot. Second, do what grandma told you to do – eat well, sleep plenty, and exercise. Third, know when to let go – more about that later.

When you are first diagnosed with cancer, it is likely to be a shock. Many cancers are diagnosed in someone without any symptoms on the basis of a test. Even when a symptom is present, the diagnosis of cancer as the underlying cause may not have been apparent. With the diagnosis, one's mind immediately goes to the questions of "How much time do I have left?" and "How good or bad will that time be?" In my case, the possibility of myeloma was identified by my cardiologist on a routine blood test as part of an annual visit to have my cholesterol-lowering statin medication adjusted. A complete blood count showed low hemoglobin, white cells, and platelets, indicating that my bone marrow was failing or being replaced by tumor or infection. I am fortunate to work at the MD Anderson Cancer Center and was evaluated later that afternoon by a hematologist friend, then referred for a bone marrow biopsy the next day. Some of the possible diagnoses had a fairly grim prognosis and others were not so bad. My mind raced through the possibilities until I received the diagnosis of myeloma with a sense of relief when the biopsy was read the next day. Of course myeloma can have serious complications such as renal injury and bone involvement, and it can have a poor prognosis depending on its genetic drivers, but it was more tractable than some of the other possibilities. I was fortunate to have a "low-risk" tumor genetic profile and no renal or bone complications and began standard-of-care treatment almost immediately.

My first piece of advice is to read. This is advice you are unlikely to need since you will be able to think of almost nothing else initially, and will be intensely interested in treatment options, prognosis, etc. Working as a pulmonologist in a cancer hospital, I had a superficial understanding of myeloma, particularly since the focus of my practice is on pulmonary complications (such as pneumonia, bronchiectasis, and obliterative bronchiolitis) of hematologic malignancies (including leukemia, lymphoma, and myeloma). Nonetheless, I began to read deeply both the classical and cutting-edge literature of myeloma. I will never know as much as my oncologist, who has spent her entire career focused on myeloma, but I think my reading served both me and her well. In clinic visits, she could move quickly through the basics because I now knew most of them, and we could focus on more subtle questions. I believe that most physicians appreciate a knowledgeable patient for exactly this reason and, even when challenged, appreciate the possibility of thinking through a routine issue in a new light. In a few situations, my challenges led to a change in care. After about 2 months of reading almost nothing but the myeloma literature, I settled back into a more normal schedule of reading the medical literature with a focus on pulmonary and general internal medicine, but also with a particular eye out for new developments in myeloma.

My second piece of advice is to remember what grandma told you. Hopefully you already have good habits of eating, sleeping, exercising, and balancing work and home life. If so, I suggest continuing with little change, and if not, talking to your grandma about common sense habits. There is a lot of advice on the Internet about outlandish diets that can beat back cancer, but with little or no evidence to support them. I have always eaten three conventionally balanced meals a day, with dinner routinely eaten with my family, and we didn't change anything during most of my illness. After high-dose cytotoxic chemotherapy, nausea is a problem so supplementation with commercial nutritional drinks may be necessary transiently, but sticking for the most part with a routine balanced diet preserves one of life's pleasures and a foundation of good health. Sleep is similar - most of us know how beneficial a full night of sleep is and what the main components of good sleep hygiene are. Some excess fatigue resulting from cancer therapy is common, so a nap can be helpful. Possibly the most important area of lifestyle management during cancer treatment is exercise. Of course this is important throughout life, but cancer and its treatment can lead to debility and frailty, so every effort should be made to prevent that. Besides, it is my impression that regular exercise minimizes symptoms from treatment and improves mood. Work is another important issue. I have always derived great pleasure from work from its intellectual interest, the companionship of colleagues and patients, and a sense of contribution. Depending on the side effects of treatment, continuing to work at either a full or reduced level may be possible, and for me it was an ongoing source of stability and pleasure. While cancer obviously imposes a strain on one's family and friends, it can also draw them closer as an unexpected benefit.

Having talked about all the things you can do to contribute to your well-being in the first and second points, I'll now pivot and remind you that you can't do it all. Clearly,

others will need to administer chemotherapy and radiotherapy and perform surgery. However, beyond that, there is a time to stop challenging the judgment of others or pushing yourself through difficult spots. For me, the most glaring example of knowing when (or when not) to let go was when I became septic during the neutropenia that followed high-dose chemotherapy and autologous hematopoietic transplantation. This was performed as an outpatient, and while it was a difficult couple of weeks, I was grateful to be in my own home with my wife as caregiver. However, late one night I developed lower abdominal pain and the urge to urinate every few minutes. I knew my neutrophil count was low, so it should have been obvious to head to the emergency room. Nonetheless, I experienced an irrational fear of a crowded emergency room in which I would end up feeling worse instead of better. When we finally went at 7 AM, I was quickly placed in a comfortable room in the emergency center; a compassionate colleague had taken a history, performed an exam, diagnosed diverticulitis, started antibiotics, and ordered placement of a Foley catheter that immediately relieved the urinary retention resulting from the diverticulitis. My experience of the rest of the hospital stay was equally caring and comfortable, and I've wondered in retrospect why I initially had such dread and delayed the obvious need for care. I suspect that part of it was due to clouded thinking from sepsis, but another part was probably an excessive desire for independence and reluctance to let go and be cared for by others. Thus, my third piece of advice is to recognize that while you can be a powerful agent in your own care, you can't do it all and it's important to trust the skill and good will of others.

In summary, I have been fortunate in my encounter with cancer, and hope that you are as well if it enters your life. I believe I did some things to aid my recovery, but also a few things that interfered, and hope you can learn from my experiences.

Editor's Note

Perhaps the most rewarding thing about editing a textbook is learning from your authors. In reading Dr. Sherry-Ann Brown's work in the area of cardio-oncologic, I was introduced to another facet of this poetic scientist's work. She has published a number of works exploring the more spiritual side of clinical practice, and two of her poems from *The Healer Speaks: Poems For Patients, Students, Doctors, Nurses, Therapists, & Everyone Impacted By Medicine* are reproduced below [3]. These poems speak to her personal relationship with cancer patients and give voice to the oftenunspoken dialogue between patient and physician (of which neither may be fully aware).

Excerpts from The Healer Speaks

Sherry-Ann Brown, MD, PhD, FACC, FAHA

Somebody Tell Me

What will kill me? When will I know? When will I die? What will I die from? Is it cancer? Is it heart disease? Is it crossing the street?! Somebody tell me!

Give me a disease of the heart. What can I do? How can I prevent this? How can I fight this? Tell me my risk of having heart disease. At least then I could walk faster, I could skip; I could jump; I could run!

> I could eat better. I could have my vegetables, My fruits, and my fish. I would even have my nuts, And extra-virgin olive oil.

I would even stop smoking. Well, I would try. Because I would want to Run away From both cancer And heart disease.

I don't know if I really could, But I surely would try. And I might even tell those around me To try too. We might even try together. Maybe as a community, We could do this together. Tell me, Tell us. What is our risk of heart disease? Tell us as a community. Tell us as a population.

> What can we do As a team To grow old together?

It's not so much about dying, As it is about living.

How is it that heart disease Kills more people In this country And in this world. Than cancer?

Then why am I more afraid of cancer?

I suppose it's because with cancer, It comes so soon!

Where did this cancer come from? She's so young. He's so young. I am so young! Why cancer? Why me? When will I die?

> What will kill me? How will I know? When will I know?

> > Oh, the agony. Breathe. Just breathe. Relax. And live. Truly live. Love to live. And give all. Till it's time.

The Healer Speaks

The Healer Speaks My dearest patient: When I see the hurt in you, I want to reach out And wrap myself around you, To protect you. I cannot myself take the pain from you, I cannot shield you from the damage Of deterioration and injury. Yet, when you hurt I hurt, When you heal I heal. You shouldn't hurt alone. I am honored to hurt with you. I am honored to hurt for you. *The Patient Speaks* My dearest doctor: Do not hurt for me. Do not hurt with me. Let me hurt without you, For my hurt is temporary, And I will heal. Whether physically, Or perhaps spiritually. I fear your hurt is irreversible, And you may not heal. While you protect me, Who will protect you? Since I cannot protect you, How will you protect yourself?

> *The Healer Responds* My dear patient: I suspect You are right.

Thank you for releasing me, To protect myself too. To care for myself too. To heal myself too. To gain healing In the midst of others too. Thank you for helping me see That first I have to take care of me.

References

- 1. Crocker PJ. Letters from the pit: stories of a physician's odyssey in emergency medicine. Cork: BookBaby; 2019. Kindle Direct Publishing, Seattle, Washington.
- Crocker PJ. More letters from the pit: stories of a physician's odyssey in emergency medicine: Kindle Direct Publishing, Seattle, Washington; 2020.
- Brown S. The healer speaks: poems for patients, students, doctors, nurses, therapists, & everyone impacted by medicine. Cheyenne: Dr Brown Cares LLC; 2019.



Correction to: Endocrinology and Metabolism

Sai-Ching Jim Yeung

Correction to: Chapter 29 in: K. H. Todd et al. (eds.), Oncologic Emergency Medicine, https://doi.org/10.1007/978-3-030-67123-5_29

Figure 29.3 had three typographical errors in the figure label in the original publication of the book. This has been replaced by a corrected version of Figure 29.3.

The updated online version of this chapter can be found at https://doi.org/10.1007/978-3-030-67123-5_29



Fig. 29.3 EKG changes in the presence of electrolyte abnormalities

Multiple-Choice Questions

- 1. Patients with the following cancers have the highest incidence (of at least one) of ED visits within 1 year of diagnosis:
 - (a) Breast, prostate, and melanoma
 - (b) Pancreas, brain, and lung
 - (c) Colon, prostate, and eye
 - (d) Oral, melanoma, and endocrine
 - (e) None of the above
- 2. Which cancers are associated with the most ED visits?
 - (a) Lung, breast, prostate, and colon
 - (b) Pancreas, brain, and lung
 - (c) Leukemias and melanoma
 - (d) Cervical, prostate, and breast
 - (e) None of the above
- 3. Limitation(s) of using ED visit-level data to explore ED use by oncologic patients generally include:
 - (a) Visit-level data may not be able to account for multiple visits made by the same patient
 - (b) Visit-level data provides no information regarding the quality of care
 - (c) Visit-level data cannot capture the incidence of ED use by all patients with cancer in a given sample or population
 - $(d) \ A \ and \ C$
 - (e) B and C
- 4. What is the most common method used to identify potentially preventable ED visits in cancer patients?
 - (a) Ambulatory care sensitive conditions
 - (b) The Billings algorithm
 - (c) Panattoni's methodology
 - (d) Prevention quality indicators
 - (e) None of the above
- 5. Which of the following statements about ED use by oncologic patient is false?
 - (a) Oncologic patients visit the ED across the cancer care trajectory.
 - (b) Underlying symptoms and the diagnoses associated with an ED visit are considered *need*-related determinants of ED use.

- (c) Oncologic patients have lower rates of admission and lower rates of multiple ED visits than the general US population.
- (d) ED use by oncologic patients reflects a complex interaction of individual and contextual factors (e.g., provider behavior, health system characteristics, and health policies).
- (e) None of the above.
- 6. Which of the following data elements is not a HIPAA identifier?
 - (a) Internet Web Address
 - (b) Birthdate
 - (c) Insurance plan number
 - (d) IP address
 - (e) Gender
- 7. What is the term for an information system that collects and analyzes data from a census of cancer cases?
 - (a) Health information exchange
 - (b) Electronic medical record
 - (c) Cancer registry
 - (d) Computerized physician order entry
- 8. Who do you want to identify first when designing an order set?
 - (a) EMR analyst
 - (b) Nursing
 - (c) Patients
 - (d) Users (i.e., physicians)
 - (e) Clinical champion
- 9. A 34-year-old man smokes a pack of cigarettes when he goes out on weekends with friends. He has smoked more than 100 cigarettes in his lifetime. This man is best classified as a:
 - (a) Current smoker
 - (b) Daily smoker
 - (c) Nonsmoker
 - (d) Insufficient information to classify smoking behavior
- 10. A 58-year-old woman with chest pain is placed in observation status in the ED's chest pain center. She smokes 30 cigarettes daily. After several hours in the
chest pain center, she becomes agitated and says she wants to walk out to the ambulance bay to smoke a cigarette. The most appropriate course of action is to:

- (a) Offer her a nicotine patch
- (b) Offer her a small dose of oral diazepam
- (c) Have her sign out against medical advice
- (d) Offer her a nicotine patch and 2 mg nicotine gum
- 11. Which of the following statements is true regarding ED-initiated tobacco control?
 - (a) Clinical trials have not demonstrated the efficacy of ED-initiated treatment.
 - (b) Nicotine replacement therapy and motivational interviewing have been shown to help ED smokers achieve abstinence.
 - (c) National bodies that accredit EDs and emergency medicine residencies require emergency care personnel to treat tobacco dependence.
 - (d) Although clinically effective, ED-initiated tobacco control is expensive.
- 12. A true statement about Screening, Brief Intervention, and Referral to Treatment (SBIRT) is:
 - (a) SBIRT was developed to facilitate the treatment of opioid-use disorder, then adapted to tobacco.
 - (b) SBIRT requires smokers to receive pharmacotherapy, such as nicotine gum or patches.
 - (c) SBIRT requires individuals with substance-use disorders to be referred to programs for post-ED care.
 - (d) SBIRT should be avoided for individuals with cooccurring serious mental illness or other substance-use disorders. These individuals should be referred to intensive outpatient programs.
- 13. When does alcohol use most commonly begin?
 - (a) Adolescence
 - (b) College-aged adults (18-22 years old)
 - (c) Middle aged adults
 - (d) Elderly
- 14. Which of the following types of cancers is *not* known to be linked to alcohol use?
 - (a) Breast
 - (b) Rectal
 - (c) Uterine
 - (d) Larynx
- 15. Which racial/ethnic group suffers from the highest rate of alcohol-attributable injury?
 - (a) Latino
 - (b) Asian
 - (c) Black
 - (d) Native American

- 16. Which diagnostic criteria for alcohol-use disorder was added to the DSM-5, from the DSM-IV?
 - (a) Tolerance
 - (b) Craving
 - (c) Legal problems
 - (d) Withdrawal
- 17. An extensively sun-damaged 60-year-old female presents after spraining her ankle. On examining the lower limb, you note a large very suspicious pigmented lesion on her calf. If you were to perform a skin biopsy to get a histological diagnosis, what would be the most appropriate?
 - (a) Punch biopsy of one edge
 - (b) Total excision of lesion with a 2-mm margin
 - (c) Shave biopsy of half the lesion
 - (d) Incisional biopsy of part of the lesion
- 18. Which of the following is *not* a high-risk melanoma phenotype?
 - (a) Red hair
 - (b) Multiple nevi (>100)
 - (c) Dark skin
 - (d) Pale eyes
- 19. What is the strongest predictor of melanoma mortality?
 - (a) Size in centimeters of the melanoma
 - (b) Breslow thickness
 - (c) Age of the person
 - (d) Age when it occurs
- 20. In the ABCDE mnemonic, which of the below is incorrect?
 - (a) Aggravating really unsightly to look at
 - (b) Border irregular
 - (c) Color multiple colors
 - (d) Diameter over the size of a pea or growing
 - (e) Evolution changing over time in size or height
- 21. How long does HPV vaccine protection last?
 - (a) 5 years
 - (b) 7 years
 - (c) 9 years
 - (d) At least 10 years
- 22. What are the US recommendations for cervical cancer screening on age of starting and method?
 - (a) At the age of sexual debut, cervical cytology
 - (b) At the age of sexual debut, cervical cytology and HPV testing
 - (c) 21 years, cervical cytology
 - (d) 21 years, cervical cytology and HPV testing
- 23. In the United States, what types of treatments are recommended for CIN2/CIN3?
 - (a) Observation

- (b) Ablation
- (c) Excisional procedure (e.g., LEEP or CKC)
- (d) Hysterectomy
- 24. What will happen with most women who become infected with HPV?
 - (a) Develop cervical cancer in 10 years
 - (b) Develop significant preinvasive disease in 5 years
 - (c) Have persistent HPV infection
 - (d) Clear the infection spontaneously within 18–24 months
- 25. A 65-year-old male presents to the ED with several days of progressive diffuse, colicky abdominal pain, with nausea, vomiting, and abdominal distension. He has been obstipated for several days and denies any rectal bleeding. He was diagnosed with ulcerative colitis and pancolitis 25 years ago. After his initial diagnosis of ulcerative colitis, he was maintained on chronic mesalamine therapy with infrequent flares of disease requiring short courses of steroids. He moved into the area 10 years ago and failed to establish contact with a local gastroenterologist. His last colonoscopy was over 10 years ago. In the ED, he appeared pale with a blood pressure of 95/60 mmHg, a heart rate of 110/minute, and temperature of 100.5 °F. Physical examination revealed a thin male in mild distress. On abdominal examination, he demonstrated a distended abdomen with diffuse tenderness and absent bowel sounds. Laboratory results show a mild lactic acidosis and decreased hemoglobin with normal LFTs. Plain films reveal a diffusely distended colon and small bowel with a point of transition in the sigmoid colon and absence of air distally. What is the most likely cause for this patient's diagnosis?
 - (a) Toxic megacolon
 - (b) Acute pancreatitis
 - (c) Ascending cholangitis with underlying primary sclerosing cholangitis
 - (d) Colon cancer
 - (e) Acute diverticulitis
- 26. The following symptoms of colorectal cancer are often suggestive of advanced disease, with the exception of:(a) Large bowel obstruction
 - (a) Large bower obstr
 - (b) Perforation
 - (c) Hematochezia
 - (d) Pneumaturia
 - (e) Abdominal pain or perirectal pain
- 27. A 53-year-old man presents to the ED with complaints of constipation and occasional blood in his stool. He has been told he has hemorrhoids in the past. Upon further questioning, he admits to a 10-pound weight loss

and constipation with intermittent diarrhea. His only medication is aspirin and HCTZ. Orthostatics in the ED are negative and his rectal examination is positive for blood. Abdominal films reveal a preponderance of stool. Hgb is 11.5 with a normal MCV. What is the best next step?

- (a) Admit for further workup
- (b) Discontinue aspirin
- (c) CT colonography
- (d) Consult GI for outpatient colonoscopy
- 28. A 74-year-old woman with a recent diagnosis of colon cancer presents to the ED after being found by her daughter on the kitchen floor. The patient states she was eating and suddenly developed abdominal pain, became lightheaded, and fell trying to get to the telephone. Hip films in the ED reveal a right trochanteric fracture. On physical examination, the patient complains of right hip and right lower abdominal pain, and she appears to be guarding on her abdominal examination. Orthopedic surgery has evaluated the patient and want clearance to take her to the OR. What should the next step be?
 - (a) Obtain a cardiac risk assessment
 - (b) Upright abdominal series
 - (c) Larger bore IV and fluid resuscitation
 - (d) Oncologic evaluation
- 29. A 68-year-old male presents to the ED with large volume hemoptysis and respiratory distress about 18 months after radiation therapy for an early-stage primary lung cancer of the right upper lobe. A CT chest scan is obtained and shows a right upper lobe opacity unchanged from previous imaging and consistent with typical postradiation changes. No other airway abnormalities are identified. What is the best next step?
 - (a) Bronchoscopy to identify and isolate the bleed
 - (b) Bronchial artery embolization
 - (c) Thoracic surgery consultation for lobectomy
 - (d) Intubation, followed by bronchoscopy to identify and isolate the bleed
- 30. A 75-year-old male smoker presents to the pulmonologist with dyspnea during minimal exertion for 4 weeks. Spirometry showed reduced peak expiratory flow rates and moderate obstruction. A CT chest without contrast showed a large, tumor obstructing the left mainstem bronchus and causing collapse of the entire left lower lobe. When considering therapeutic bronchoscopy, what are the main features associated with improvement in symptoms?
 - (a) Greater dyspnea, good functional status
 - (b) Greater dyspnea, poorer functional status

- (c) Minimal dyspnea, poorer functional status
- (d) Minimal dyspnea, good functional status
- 31. An 81-year-old male with a history of advanced stage primary lung cancer of the right lower lobe treated with chemotherapy and radiotherapy presents for surveillance 1 year after treatment. On review of the CT, there appears to be a local recurrence of the tumor with growth into the right bronchus intermedius and complete collapse of right lower lobe. The tumor has completely invaded and destroyed the right lower lobe bronchi. Which one of the following factors is a predictor of successful therapeutic bronchoscopy?
 - (a) Presence or absence of necrosis
 - (b) Size and extent of endobronchial invasion
 - (c) Patency of the lobar bronchi distal to the obstruction
 - (d) Type of lesion (intraluminal, extraluminal, mixed)
- 32. A 62-year-old woman is undergoing rigid bronchoscopy under anesthesia for a bleeding tumor in the left mainstem. While the bronchoscopist is coagulating the surface of the tumor with argon plasma coagulation, the patient suddenly develops ventricular fibrillation and arrest. What is the rare but dreaded complication that could have caused this event?
 - (a) Idiosyncratic reaction to intravenous anesthetics
 - (b) Gas embolism from argon plasma coagulation
 - (c) Perforation of the left atrium
 - (d) Myocardial infarction
- 33. A 49-year-old man with retroperitoneal sarcoma on chemotherapy is in the ED with a 3-day history of abdominal pain, distension, and fever of 38.5 °C. His lactic acid is 4 mg/dL, and the blood pressure is 70/50 mmHg. He takes extended release morphine, 60 mg q12h and immediate release morphine. His absolute neutrophil count is $100/\mu$ L. He denies confusion. The triage nurse found that he was alert, and oriented x 3. His wife says he sleeps more and is withdrawn. She has not noticed agitation or hallucinations. You decide to reevaluate his mental status with the mini-mental status examination (MMSE). Which domain(s) of this test are most helpful to confirm hypoactive delirium?
 - (a) Orientation to time, place, and self.
 - (b) No specific domain is better than the other.
 - (c) The test will be unremarkable so the patient has no delirium.
 - (d) Attention/concentration (backward spelling or serial 7 subtractions and 5-min memory recall of three phrases).
- 34. A 62-year-old man with recent diagnosis of esophageal adenocarcinoma has arrived in the ED with acute onset garbled, nonsensical speech. Time of onset was 1 hour

before your bedside examination, which reveals aphasia and right pronator drift. There is no history of atrial fibrillation, seizure, or migraine. CT of head shows no intracranial hemorrhage (ICH) or areas of vasogenic edema. EKG reveals sinus rhythm. His BP is 145/80 mmHg. You determine that the patient probably has an acute ischemic stroke in the left middle cerebral artery or one of its branches. There is no evidence of thrombocytopenia, active bleeding, or abnormal coagulation tests. The most important therapeutic decision at this point is to:

- (a) Admit to the floor on telemetry and consult neurology to see later.
- (b) Give ASA 81 mg stat and order MRI brain to confirm diagnosis.
- (c) After a quick checklist to ensure there is no absolute contraindication and discussion with patient or proxy for consent, start IV thrombolytics or transfer to a dedicated stroke center for thrombolytic therapy.
- (d) Start intravenous heparin and transfer to ICU.
- 35. Paramedics brought a 26-year-old woman who was found unresponsive in her bedroom. She has acute myeloid leukemia in relapse. Her mother said she had been complaining of headaches for 2 days. There are no signs of intentional or accidental drug overdose and no history of illicit drug use or suicidal attempts. She was intubated on the spot for airway protection. Her Glasgow Coma Scale score is 9, pupils measure 3 mm and both react to light, and bilateral Babinski reflexes are present. The paramedics report she has been seizing despite repeat doses of lorazepam (total of 4 mg). You witness the patient hyperventilate and arch her body, while the blood pressure rises from 100/75 to 190/110 mmHg, with bradycardia. Her pupils dilate for 15 seconds. She then returns to baseline. CT scan of brain showed an epidural hematoma with mass effect and acute hydrocephalus. This episode is more likely to represent:
 - (a) Convulsive status, refractory to benzodiazepines
 - (b) Nonconvulsive epileptic status
 - (c) Plateau wave
 - (d) None of the above
- 36. A 34-year-old woman with acute myeloid leukemia in relapse was found unresponsive. She had complained of headaches for 1 week. She was intubated and her Glasgow Coma Scale score is 9, pupils measure 3 mm and both react to light, and bilateral Babinski reflexes are present. She has been seizing despite repeat doses of lorazepam (total of 4 mg). You witness the patient hyperventilate and arch her body, while the blood pressure rises from 100/75 to 190/110 mmHg, with brady-

cardia. Her pupils dilate for 15 seconds. She then returns to baseline. CT scan of brain showed an epidural hematoma with mass effect and acute hydrocephalus. Laboratory studies reveal a platelet count is $45,000/\mu$ L and INR 1.25. The most important treatment sequence is:

- (a) Start osmotherapy, replace coagulation factors, and emergently consult neurosurgery for hematoma evacuation.
- (b) Supportive care only, get an EEG and transfer to ICU with antiepileptic drug therapy.
- (c) Transfer to ICU and start osmotherapy. Consult neurosurgery to see later.
- (d) Keep in the ED, start osmotherapy with mannitol or hypertonic saline and consult neurosurgery.
- 37. A 64-year-old female with metastatic breast cancer presents to the ED with new back pain. The patient is ambulatory, but this week finds herself holding on to furniture or using a cane when she walks. Physical assessment reveals spinal tenderness in the L2–L3 vertebral area. X-ray findings identify osteoarthritic changes, but no other significant abnormalities. What is the next appropriate step for patient evaluation and management?
 - (a) Obtain CT spine
 - (b) Obtain MRI spine
 - (c) Refer to outpatient pain management services
 - (d) Administer zoledronic acid (zoledronate)
- 38. Upon review of MRI imaging, a patient is found to have a vertebral body metastasis on the T8 vertebral body. The imaging identifies a deformation of thecal sac without cord abutment. What is the grade of epidural disease based on Bilsky's grading system, and what is the likely treatment for the finding?
 - (a) Grade 1C; radiation therapy
 - (b) Grade 2; radiation therapy
 - (c) Grade 1B; radiation therapy
 - (d) Grade 1C; long-term corticosteroids
- 39. An 84-year-old male with prostate cancer presents with new-onset back pain, hyperreflexia, bilateral weakness in lower extremities, and urinary incontinence. The clinician has a high suspicion of metastatic spinal cord compression (MSCC), which is confirmed via MRI. Which finding provides indication of poor prognosis for this patient?
 - (a) Presence of MSCC
 - (b) Bilateral weakness in lower extremities
 - (c) Hyperreflexia
 - (d) Urinary incontinence
- 40. When determining medical treatment for identified metastatic spinal cord compression (MSCC), all of the

following patients would be appropriate candidates for corticosteroid therapy, except:

- (a) A 66-year-old male with metastatic lung cancer
- (b) A 58-year-old female with metastatic breast cancer
- (c) A 77-year-old male with diagnosed prostate cancer
- (d) A 33-year-old male without a cancer diagnosis, presenting with back pain and scrotal edema
- 41. Which hormonal profile of endocrine disturbance is most common in a patient with pituitary apoplexy?
 - (a) Low cortisol, high prolactin, and diabetes insipidus.
 - (b) Low cortisol, diabetes insipidus, and other pituitary hormones normal.
 - (c) Low cortisol, no diabetes insipidus, and low growth hormone.
 - (d) Normal cortisol, diabetes mellitus, and low TSH.
 - (e) No hormonal disturbance is present.
- 42. The most common cranial neuropathy causing diplopia in patient with pituitary apoplexy is:
 - (a) IV nerve (trochlear nerve) palsy
 - (b) V nerve (trigeminal nerve) palsy
 - (c) IX nerve (glossopharyngeal nerve) palsy
 - (d) III nerve (oculomotor nerve) palsy
 - (e) II nerve (optic nerve) dysfunction
- 43. Hemorrhage within a pituitary tumor on an MRI of the sella is best conformed by the presence of which of the following features?
 - (a) Isointense on T1-weighted images for the first 3 days, turning hyperintense thereafter
 - (b) Hyperintense on T1-weighted images for the first 24 hours, turning hypointense until day 7, then turning hyperintense again
 - (c) Hyperintense on T2-weighted images until day 7, turning hypointense thereafter
 - (d) Isointense on T1-weighted images for the first 3 days, turning hypointense thereafter
 - (e) Hypointense on T1-weighted images both before and after gadolinium contrast is given
- 44. Which of the following is *not* a factor predisposing patients to develop pituitary apoplexy?
 - (a) Coagulopathy
 - (b) Hypertension
 - (c) Use of rivaroxaban
 - (d) Pregnancy
 - (e) Use of a dopamine antagonist
- 45. A 65-year-old male heavy smoker presents with 6 weeks of progressively worsening hoarseness, dysphagia, and dyspnea with a 20-pound weight loss over that period. He is seated in the tripod position and has

biphasic stridor on examination. What is the most appropriate next step in management while awaiting an otolaryngology consult?

- (a) CT neck soft tissue with contrast
- (b) Emergent intubation in the ED
- (c) Intravenous steroids
- (d) Intravenous saline bolus
- 46. A 74-year-old male with a history of advanced squamous cell carcinoma of the base of tongue treated with concurrent chemoradiation 6 years prior presents with acute bleeding from the mouth this morning. He had been at work when he had brisk bleeding from his mouth which resolved spontaneously after approximately 5 minutes. His vitals are within normal limits other than mild tachycardia and his hemoglobin is 8.7 g/dl. He is no longer having any active bleeding and is breathing comfortably on room air. What is the next appropriate step in management?
 - (a) Urgent intubation
 - (b) CT angiogram of head and neck
 - (c) GI consultation
 - (d) Discharge with close outpatient follow-up
- 47. A 62-year-old male nonsmoker presents to the ED with an enlarging right neck mass over the past few months. He has not noticed any dysphagia, odynophagia, voice changes, or dyspnea. His vital signs are normal. On CT neck, he is noted to have a 3 cm cystic mass of the right neck. What is the most likely diagnosis?
 - (a) Branchial cleft cyst
 - (b) Lymphoma
 - (c) Abscess
 - (d) Metastatic squamous cell carcinoma
- 48. Which of the following features on presentation is most suggestive of sinonasal malignancy rather than acute bacterial sinusitis?
 - (a) Bilateral nasal congestion
 - (b) Facial pain and pressure for 14 days
 - (c) Facial paresthesia
 - (d) Purulent nasal drainage
- 49. A 48-year-old female presents with lower extremity claudication several months after beginning therapy with a new medication for acute lymphocytic leukemia. Which is the most likely medication responsible for her premature peripheral artery disease?
 - (a) Doxorubicin
 - (b) Trastuzumab
 - (c) Ponatinib
 - (d) Ibrutinib
- 50. A 53-year-old male presents with dyspnea and chest discomfort 28 days after initiation of pembrolizumab

to treat non-small cell lung cancer. Which of the following is true regarding myocarditis induced by immune checkpoint inhibitors?

- (a) Troponin is always abnormal.
- (b) Symptoms usually appear within the first few months of therapy.
- (c) Cardiac MRI is always required for diagnosis.
- (d) ECG never appears normal.
- 51. A 74-year-old female is hospitalized with new systolic heart failure 10 years after doxorubicin therapy for breast cancer. Which of the following options have been suggested for prevention of anthracycline-induced cardiac dysfunction?
 - (a) Angiotensin-converting enzyme inhibitors
 - (b) Beta blockers
 - (c) Dexrazoxane
 - (d) All of the above
- 52. A 62-year-old male is admitted with stress cardiomy-opathy following infusion with 5-fluorouracil for rectal adenocarcinoma. When is stress cardiomyopathy most commonly reported to associate with chemotherapy?(a) Initial chemotherapy infusion
 - (a) Initial chemotherapy infusion
 - (b) Last chemotherapy infusion
 - (c) Second chemotherapy infusion
 - (d) Mid-cycle chemotherapy infusion
- 53. A 56-year-old male with a history of non–small cell lung cancer presents with worsening shortness of breath due to a large loculated pleural effusion. He had undergone three thoracenteses recently for symptomatic recurrent pleural effusions. The last thoracentesis was aborted early due to chest pressure after draining 300 ml of pleural fluid. Pleural studies reveal lymphocytic rich exudative with negative cytology. What is the best next step?
 - (a) Repeat therapeutic thoracentesis.
 - (b) Perform pleuroscopy/medical thoracoscopy, pleural biopsy, and talc pleurodesis.
 - (c) Insert a chest tube and perform chemical pleurodesis.
 - (d) Perform pleuroscopy, pleural biopsy and insert an indwelling pleural catheter.
- 54. A 78-year-old female with a history of coronary artery disease, diabetes, and HFrEF of 35% underwent an AICD placement successfully and was transferred to the recovery room. While patient was in the recovery room, he developed acute onset dyspnea and left-sided chest pressure requiring supplemental oxygen. His SPO₂ was varying between 85% and 89% on 4 liters of supplemental oxygen. Stat chest X-ray at the bedside showed a moderate pneumothorax on the left side.

Patient undergoes a left chest tube placement with improvement in symptoms. Which statement is correct regarding the management of pneumothorax?

- (a) Large chest tube (20–24 F) helps in resolution of pneumothorax.
- (b) There is no evidence that large chest tubes (20–24 F) are better than small bore chest tubes (10–14 F) in the management of pneumothoraces.
- (c) Positive pressure helps resolve pneumothoraces rapidly.
- (d) Endobronchial valves have no role in the treatment of persistent air leaks from pneumothorax.
- 55. Which statement regarding permissive hypercapnia in patients with ARDS using low tidal volume ventilation is incorrect?
 - (a) Rise of PaCo₂ increases tissue oxygenation by right shift of oxygen-hemoglobin dissociation curve.
 - (b) Hypercapnic acidosis increases cyclic mechanical stretch, epithelial injury, and cell death.
 - (c) Hypercapnic acidosis increases epithelial injury and cell death compared to normocapnia.
 - (d) Hypercapnic acidosis reduces cyclic mechanical stretch, epithelial injury, and cell death.
- 56. A 68-year-old MALE with history of squamous cell lung carcinoma presented to ED with complains of moderate hemoptysis (30–40 cc) for the past 5 days. He remains hemodynamically stable with oxygen saturation of >95% but continues to have cough blood clots. Which of the following would be the best approach to the treatment?
 - (a) Nebulized tranexamic acid (TXA)
 - (b) Cough suppressants and supportive care
 - (c) Emergent surgical consultation
 - (d) Therapeutic bronchoscopy
- 57. A 34-year-old homeless female with stage IV breast cancer, undergoing doxorubicin treatment presents to the ED with a swollen right lower extremity, with ery-thema of the calf and foot, and exquisite tenderness to right calf. What test should be ordered, and once the diagnosis is made what would be the appropriate therapy?
 - (a) D-dimer, if positive, start a TSA (target specific anticoagulation) and follow-up with oncologist as scheduled.
 - (b) DVT ultrasound, if positive DVT, CTPA of the chest to rule out pulmonary embolism, and then initiate lovenox at 1 mg/kg to bridge to vitamin K antagonist. Ultimately discharge patient home with follow-up.

- (c) DVT ultrasound, if positive for DVT, admit to the hospital for further workup and treatment.
- (d) DVT ultrasound, if positive for DVT, discharge with target-specific anticoagulation (TSA).
- 58. A 65-year-old male with diabetes and hypertension, as well as a new diagnosis of pancreatic cancer, presents to the ED with shortness of breath. He looks ill, panicking, visibly short of breath, with a blood pressure of 85/45, heart rate of 120 beats per minute, SpO₂ of 88% on room air, SpO₂ of 94% on 6 L nasal cannula, and a temperature of 37 °C. STAT laboratory results demonstrate an elevated troponin and elevated pro-BNP, and bedside ultrasound demonstrates an enlarged right ventricle without an effusion. What is the next step in management of this patient?
 - (a) CTPA of the chest to confirm diagnosis of pulmonary embolism
 - (b) Obtain D-Dimer, if elevated, CTPA of the chest to confirm pulmonary embolism
 - (c) Interventional radiology consultation of direct thrombolysis
 - (d) Systemic fibrinolysis
- 59. A 66-year-old female with a history of non-small cell lung cancer is sent to the ED after being told she has a pulmonary embolism seen incidentally on a routine follow-up chest CT scan. It is a single subsegmental pulmonary embolism with no evidence of right heart strain on CT scan. She also has no leg swelling or calf pain. Her vital signs are within normal limits and she has no complaints. What is the best next step in management of this pulmonary embolism?
 - (a) Since it was an incidental finding, discharge without patient follow-up.
 - (b) Obtain basic laboratory results, initiate intravenous heparin infusion and admit to telemetry.
 - (c) Patient is Hestia, high risk for outpatient management. Start lovenox in the ED, so the admitting team can bridge to a vitamin K antagonist.
 - (d) Obtain basic laboratory results, if creatinine clearance is normal discharge on rivaroxaban and follow-up with the oncologist in 1 week.
- 60. A 65-year-old male with hyperlipidemia, hypertension, and a 50-pack year smoking history presents with unilateral left lower extremity swelling for 4 days. He has associated left calf tenderness, and tenderness along the left deep venous system. He had a cough a few weeks ago with some blood tinged sputum but that has resolved. His vital signs are normal, and he has obvious left lower extremity swelling and some redness. Homan's sign is positive, D-dimer is elevated,

and subsequent left leg duplex ultrasonography demonstrates partially occlusive thrombosis of the popliteal vein. A diagnosis of DVT is made. What is the best next step in this patient's care?

- (a) Obtain CTPA of chest to evaluate for lung cancer, and if positive, admit to the hospital for workup of his cancer and initiation of anticoagulation.
- (b) Discharge home on anticoagulation and inform the patient that he must follow up with his primary care physician to be evaluated for malignancy as well as additional other causes of VTE.
- (c) First-time DVTs do not require any further evaluation; the patient can complete 3 months of anticoagulation and follow-up as needed.
- (d) Obtain CTPA to evaluate for pulmonary embolism, if positive, consult interventional radiology for IVC filter and admit the patient on lovenox for bridging to vitamin K antagonist.
- 61. A 62-year-old man with squamous cell carcinoma of the lung presents to the ED with weakness, nausea, and shortness of breath. His cancer is stage IV, and he has had three cycles of treatment with pembrolizumab, carboplatin, and paclitaxel. He had been a long-time smoker and has chronic obstructive pulmonary disease (COPD). In the ED, he is tachypneic with a respiratory rate of 30/min. He is not hypoxic on room air by pulse oximetry. His vital signs are otherwise normal. His physical examination is normal except for signs of dehydration. What is the differential diagnosis?
 - (a) Diabetic ketoacidosis
 - (b) Pneumonia (bacterial or viral, including COVID-19)
 - (c) Immune pneumonitis
 - (d) Pulmonary embolism
 - (e) COPD exacerbation
 - (f) B, C, D, and E
 - (g) All of the above
- 62. A 65-year-old man with advanced melanoma has received 2 cycles of the combination of nivolumab and ipilimumab, and now presents to the ED with head-ache, general weakness (too weak to get out of bed), nausea, and loss of appetite. He was brought in by ambulance. On arrival, his vital signs are: BP 89/40 mmHg, pulse 89/minute, respiratory rate 18/ minute, oxygen saturation on room air 96%, and temperature 96.5 °F. He is listless but easily arousable and is oriented to time, place, and person. Initial supportive care is initiated including intravenous normal saline bolus. His complete blood count with differential is normal, but the comprehensive metabolic panel shows a serum sodium of 128 mEq/L, chloride of 93 mEq/L,

and bicarbonate of 28 mEq/L. His electrocardiogram shows no ST segment changes suggestive of myocardial ischemia. What is the next step(s) to diagnose the cause of his symptoms?

- (a) Stat random cortisol level and adrenocorticotropin level
- (b) TSH and free thyroxine
- (c) MRI of the sella
- (d) All of the above
- 63. A 36-year-old woman with papillary thyroid carcinoma presents to the ED with numbness and tingling of hands and feet. She was discharged home from the hospital 2 days ago after she underwent total thyroidectomy 4 days ago. She complains that her left calf is painful after she had a leg cramp when she woke up that morning. Her vital signs are normal. Her surgical wound is within normal limits of a postoperative status. No hypoxia. No chest pain. No dyspnea. No edema. What is the most likely diagnosis?
 - (a) Deep venous thrombosis in the left leg
 - (b) Anxiety attack and hyperventilation causing numbress and tingling of hands and feet
 - (c) Hypothyroid symptoms due to postsurgical hypothyroidism
 - (d) Postoperative hypoparathyroidism
- 64. A 55-year-old woman with metastatic colon cancer has had her 3 cycles of chemotherapy with panitumumab, 5-fluorouracil, and oxaliplatin 4 days ago. She presents to the ED for general weakness and muscle cramping. She is irritable and has been crying. Her CBC-diff showed mildly depressed blood counts that do not require transfusion. Her complete metabolic panel shows sodium 133 mEq/L, potassium 3.3 mEq/L, calcium 8.0 mg/dL, and magnesium 0.3 mg/dL. Which of the following statement is true?
 - (a) Stat EKG, and monitor cardiac rhythm while electrolyte abnormalities are corrected with intravenous infusions since the patient is at high risk for cardiac arrhythmia.
 - (b) Hypomagnesemia causes PR shortening and thereby increases the risk for torsade de pointes.
 - (c) Hypomagnesemia is caused by the nephrotoxic effect of oxaliplatin and is unrelated to 5-fluorouracil and panitumumab.
 - (d) All of the above are true.
- 65. One of the following is the least effective endoscopic hemostatic modalities for GI bleeding related to tumors:
 - (a) Cryotherapy
 - (b) Argon plasma coagulation
 - (c) Nd-YAG laser

- (d) Endoscopic clips
- (e) Hemostatic powders (when available on the market)
- 66. In patients who need palliation of their symptoms due to duodenal obstruction from pancreatic cancer, an enteral stent should be placed if their life expectancy is expected to be greater than 6–12 months.
 - (a) True
 - (b) False
- 67. All of the following are potential causes of fulminant hepatic failure in a patient with cancer, except:
 - (a) Chemotherapy
 - (b) Innumerable hepatic metastases
 - (c) Total parenteral nutrition
 - (d) Malignant infiltration of the liver with lymphoma or adenocarcinoma
 - (e) Polypharmacy/drug-induced liver injury
- 68. Which of the following is *not* an indication for ERCP for biliary decompression?
 - (a) Intolerable pruritus
 - (b) Intolerable jaundice
 - (c) Acute cholangitis
 - (d) Abnormal LFTs without symptoms
 - (e) Hyperbilirubinemia that interferes with chemotherapy dosing
- 69. Current management of adhesive small bowel obstruction includes nonoperative management for up to 3 days, as long as the patient does not deteriorate with symptoms suggesting intestinal ischemia.
 - (a) True
 - (b) False
- 70. Advanced malignant disease in the upper gastrointestinal tract can result in gastric outlet obstruction. The most common malignant cause of gastric outlet obstruction is:
 - (a) Pancreatic cancer
 - (b) Periampullary cancer
 - (c) Advanced gastric cancer
 - (d) Duodenal and jejunal cancer
- The classic presentation of neutropenic enterocolitis is a patient with absolute neutrophil count <1000 cells/ μL, new-onset abdominal pain, and fever. Most commonly it occurs:
 - (a) 1 week after receiving cytotoxic chemotherapy
 - (b) 2–3 weeks after receiving cytotoxic chemotherapy
 - (c) 4 weeks after receiving cytotoxic chemotherapy
- 72. Acute radiation enteritis usually occurs between 2 and 6 weeks postradiation therapy. Clinical manifestations include nausea, vomiting, abdominal pain, and diar-

rhea associated with blood and mucus. Possible complications include:

- (a) Systemic sepsis
- (b) Gastrointestinal bleeding
- (c) Bowel obstruction
- (d) All of the above
- 73. A 30-year-old woman with colon cancer and Lynch syndrome. She had metastatic colon cancer that has started to progress after first-line cytotoxic chemotherapy. She has just received the third cycle of ipilimumab and nivolumab 3 days ago. She presents to the ED with severe watery diarrhea. She had more than 8 bowel movements in the past 24 hours. Her blood pressure is normal, but she is tachycardic. Her bowel sound is hyperactive. Her right upper quadrant tenderness is unchanged. Which of the following are true?
 - (a) The patient may have grade 3 immune-mediated colitis, and CT scan of the abdomen and pelvis is indicated.
 - (b) Call the gastroenterology consult team to arrange for colonoscopy and biopsy.
 - (c) Collect stool samples for culture and various other tests for infectious diarrhea (including *C. difficile* toxins).
 - (d) All of the above.
- 74. A 65-year-old patient with advanced VIPoma currently on long-acting pegylated octreotide now presents to the ED with worsening of chronic diarrhea. Over the last week, his stool frequency gradually increased to 6 times per day. He feels weak and dehydrated. Which of the following is true?
 - (a) This is a chronic problem, and the patient can be discharged from the ED after intravenous hydration with isotonic crystalloid solutions and replacement of electrolytes.
 - (b) Intravenous octreotide is not helpful since the patient is already on long-acting octreotide injections.
 - (c) Consult interventional radiology to evaluate whether reduction of tumor burden by chemoembolization is possible.
 - (d) None of the above.
- 75. A 55-year-old man with acute myeloid leukemia, who had allogeneic hematopoietic stem cell transplant 3 weeks ago, presents to the ED with severe diarrhea, nausea, and vomiting for 2 days. He has had 10 watery large volume bowel movements in the past 24 hours despite taking loperamide around the clock. What is the appropriate management if he is hypotensive, tachycardic, and hypokalemic?

- (a) Obtain blood cultures, urine culture, and urinalysis; initiate intravenous fluids 30 mL/kg; and start broad-spectrum empirical antibiotics.
- (b) Collect stool sample for *C. difficile* toxin assay, multiplex PCR for gastrointestinal pathogens, and stool culture.
- (c) Discuss with the stem cell transplant team to consider starting intravenous glucocorticoid therapy.
- (d) All of the above.
- 76. A 22-year-old woman with acute lymphoblastic leukemia, s/p 2 cycles of hyper-CVAD chemotherapy presents to the ED for sudden onset fever, chills, and mild epigastric abdominal pain. She had a CBC-diff the day before, and her absolute neutrophil count was 100 per microliter. She reports that she has had 2 watery bowel movements in the past 8 hours. Which is the following is true?
 - (a) CT scan of the abdomen and pelvis is indicated to diagnose neutropenic enteritis.
 - (b) Prescribe a lidocaine containing gastrointestinal cocktail to control esophageal and gastric pain from chemotherapy, and loperamide for mild chemotherapy-induced diarrhea. No further investigation of the gastrointestinal tract is needed.
 - (c) Initiate intravenous cefepime for neutropenic fever in the ED, prescribe loperamide for the mild diarrhea, and discharge the patient on home intravenous antibiotic therapy.
 - (d) None of the above.
- 77. A 56-year-old woman with glioblastoma multiforme presents to the ED with complains of not having a bowel movement for over 4 days. She has had intermittent constipation in the past and this feels similar. She also complains of nausea. Her vital signs are normal and her examination is unremarkable, except for mild dehydration. What is the best next step in management?
 - (a) Give her ondansetron for her nausea and hydrate her with intravenous fluids.
 - (b) Give her metoclopramide for her nausea and hydrate her with intravenous fluids.
 - (c) Obtain a plain radiograph to evaluate for bowel obstruction and stool burden, and hydrate her with intravenous fluids.
 - (d) Hydrate her with intravenous fluids.
- 78. A 22-year-old woman with breast cancer with metastasis to the bone presents to the ED with constipation of over 1-week duration. She has a history of chronic constipation since being diagnosed with metastatic bony disease. She is on a chronic opioid regimen. She recently increased the dose of her opioid due to uncon-

trolled cancer pain. After a thorough workup, including a digital rectal examination, you conclude that she has uncomplicated, likely opioid-induced constipation. What is the best next step in management?

- (a) Give the patient methylnaltrexone.
- (b) Give the patient senna and bisacodyl.
- (c) Encourage the patient to eat more fiber.
- (d) Disimpact the patient in the ED.
- 79. An 80-year-old male with metastatic prostate cancer to the bone presents to the ED with constipation. He has not had a bowel movement for over 1 week and is complaining of lower back pain, which he attributes to the constipation. The review of systems reveals tingling of the lower extremities. A thorough physical examination reveals mild weakness of lower extremities and bony tenderness over the spine but otherwise normal. What is the best next step in management?
 - (a) Give the patient analgesics and start IV fluid hydration, while you await laboratory results.
 - (b) Give the patient a milk and molasses enema.
 - (c) Order an emergent MRI of the spine to evaluate for malignant spinal cord compression or conus syndrome.
 - (d) Order and emergent CT of the abdomen to evaluate for intra-abdominal pathology.
- 80. A 62-year-old male presents with constipation for over 2 weeks. On arrival to the ED, he is noted to have a temperature of 38 °C (100.4 °C), a heart rate of 140 beats per minute, and blood pressure of 100/60 mmHg. He complains of diffuse abdominal pain and nausea. On physical examination, he is ill appearing and has a diffusely tender abdomen with distention and rebound tenderness. What is the most important next step in managing this patient?
 - (a) Obtain a CBC, chemistries, and a CT of the abdomen with contrast.
 - (b) Send a sepsis workup, including lactic acid.
 - (c) Give the patient an IV fluid bolus and obtain a stat upright portable chest radiograph.
 - (d) Call a surgical consult.
- 81. All of the following drugs are associated with thrombotic microangiopathy (TMA), except:
 - (a) Cisplatin
 - (b) Gemcitabine
 - (c) Anti-VEGF therapy
 - (d) Ifosfamide
- 82. All of the following are effective therapies for methotrexate (MTX) toxicity, except:
 - (a) IV hydration to increase urine output
 - (b) Bicarbonate to increase solubility of MTX in the urine

- (c) Dialysis to clear MTX
- (d) Glucarpidase to metabolize MTX into inactive compounds
- 83. All of the following may be used in the treatment of SIADH, except:
 - (a) Normal saline
 - (b) Salt tablets
 - (c) Vasopressin receptor antagonists
 - (d) Fluid restriction
- 84. Which one of the following is true about tumor lysis syndrome?
 - (a) Rasburicase may be used in the treatment of established TLS to decrease serum uric acid levels.
 - (b) Allopurinol will help to breakdown uric acid in the bloodstream.
 - (c) Preemptive dialysis is an established treatment for the prevention of TLS.
 - (d) Routine alkalization of the urine is part of the treatment of established TLS.
- 85. A 63-year-old man underwent radical prostatectomy 1 week ago. His Foley catheter was removed 36 hours ago. He initially voided well, but experienced hematuria and a weak stream after a bout of coughing. He now presents to the ED complaining of lower abdominal pain and an inability to void for 10 hours. After initial evaluation, the ED provider should:
 - (a) Proceed with an urgent CT scan of the abdomen and pelvis.
 - (b) Place a 22-French Foley and irrigate the bladder until clear.
 - (c) Place an 18-French Foley and initiate continuous bladder irrigation.
 - (d) Contact the urology service, immediately.
- 86. Continuous bladder irrigation (CBI) is indicated in the patient with persistent gross hematuria:
 - (a) If the patient passes large clots
 - (b) When there is difficulty irrigating clots from the bladder
 - (c) Once all sizeable clots have been evacuated from the bladder
 - (d) After anticoagulation has been discontinued
- 87. In a patient with ureteral obstruction and infection, the factor that triggers decompression of the urinary system with retrograde ureteral stenting versus percutaneous nephrostomy tube placement is the:
 - (a) Active use of anticoagulants
 - (b) Service line availability
 - (c) Etiology of the obstruction
 - (d) All of the above

- 88. A 75-year-old woman underwent right partial nephrectomy for a 3 cm renal carcinoma. One week later she presents to the ED with gross hematuria and right flank pain. She is hemodynamically stable with a normal WBC, creatinine, and a hemoglobin of 11 g/dl. The next step is to:
 - (a) Discharge home with follow-up within 1 week.
 - (b) Monitor the patient in the ICU.
 - (c) Proceed with a CT angiogram.
 - (d) Obtain a CT urogram and urine cytology.
- 89. A 68-year-old female recently diagnosed with metastatic ovarian cancer was found to be confused with muscle weakness and diminished reflexes. Which laboratory abnormality would be the cause of the symptoms?
 - (a) Hypocalcemia
 - (b) Hypercalcemia
 - (c) Hyperchloremia
 - (d) Hypochloremia
- 90. A 45-year-old female who has been undergoing carboplatin and paclitaxel for ovarian cancer presents with a temperature of 38.0 °C for at least an hour. What other laboratory abnormality would you need to diagnose febrile neutropenia?
 - (a) ANC $\leq 500 \text{ cells/mm}^3$
 - (b) ANC \geq 500 cells/mm³
 - (c) ANC $\leq 300 \text{ cells/mm}^3$
 - (d) ANC $\leq 400 \text{ cells/mm}^3$
- 91. Which of the following is *not* a noninvasive method for control of bleeding in a patient with advanced cervical cancer?
 - (a) Palliative radiotherapy
 - (b) Vaginal packing
 - (c) Percutaneous uterine artery embolization
 - (d) Volume replacement
- 92. A 58-year-old patient with a history of ovarian cancer treated with intraperitoneal cisplatin and paclitaxel presents to the ED with vomiting, constipation, abdominal distension, and abdominal pain. CT scan shows bowel thickening and mesenteric stranding. The patient's temperature is within normal limits. What would be the most likely diagnosis?
 - (a) Diverticulitis
 - (b) Intestinal obstruction
 - (c) Chemotherapy-induced enterocolitis
 - (d) Pelvic inflammatory disease
- 93. A 72-year-old female with history of hypertension and a 30-pack year smoking history presents with left leg pain and inability to ambulate. She has had pain in her

mid-left tibia for the past 4 weeks and while walking to the restroom this morning, felt a pop with associated deformity. On physical examination, her left leg is grossly deformed, but the skin is intact. Her compartments are soft without associated numbness or tingling, and distal pulses remain palpable. Radiographs show a large lytic lesion in the tibial diaphysis with a displaced pathologic fracture. What is the most likely diagnosis, and how should additional workup in the ED proceed?

- (a) Primary bone sarcoma. CT of the chest, abdomen, and pelvis with laboratory workup assessing for myeloma and primary sarcoma. Leg should be immobilized in a short leg splint.
- (b) Primary bone sarcoma. CT of the chest, abdomen, and pelvis with laboratory workup assessing for myeloma and primary sarcoma. Leg should be immobilized in a long leg splint.
- (c) Metastatic disease. CT of the chest, abdomen, and pelvis with laboratory workup assessing for myeloma and metastatic disease. Leg should be immobilized in a short leg splint.
- (d) Metastatic disease. CT of the chest, abdomen, and pelvis with laboratory workup assessing for myeloma and metastatic disease. Leg should be immobilized in a long leg splint.
- 94. A 65-year-old male with history of metastatic prostate cancer to the left femoral neck, 8 weeks status post left total hip replacement, presents to the ED with inability to bear weight. He was standing up after using the toilet earlier this morning and felt a sudden pop in his left hip. He is neurovascularly intact. Laboratory results are not concerning for infection. What are radiographs likely to show? What is the most likely position of his leg on physical exam? What is the most appropriate management in the ED?
 - (a) Posterior dislocation of left total hip replacement.
 Shortened, flexed, adducted, internally rotated.
 Conscious sedation and closed reduction
 - (b) Anterior dislocation of left total hip replacement. Shortened, flexed, abducted, externally rotated. Conscious sedation and closed reduction
 - (c) Vancouver B periprosthetic fracture. Shortened, slightly externally rotated. Buck's traction
 - (d) Vancouver C periprosthetic fracture. Shortened, slightly externally rotated. Buck's traction
- 95. A 48-year-old female with history of diabetes and softtissue sarcoma in her left lateral thigh, who received preoperative radiation and is now 3 weeks status post radical resection, presents to the ED with her son, who status she "doesn't seem like herself." On examination, she has marked erythema and induration about her sur-

gical incision with multiple hemorrhagic bullae. There is a focal area of wound dehiscence distally with "dishwater" fluid draining. She is oriented to person, but not place or time. She has a WBC of 34,000/mm³, hemo-globin of 9.8 g/dL, sodium of 128 mmol/L, and CRP of 320 mg/L. CT does not show any evidence of gas. What is the most likely diagnosis and what is the most appropriate next step?

- (a) Cellulitis. Empirical antibiotics and potentially surgery
- (b) Abscess. Admission to the hospital and surgery within the next 24–48 hours
- (c) Necrotizing fasciitis. Emergent surgical consult and surgery
- (d) Suture reaction. Oral antibiotics and discharge home
- 96. A 68-year-old male with acute myelogenous leukemia on chemotherapy presents with acute onset pain and swelling in his right leg. Pain started earlier this morning. He states he may have bumped his leg on the dresser, but he does not recall any significant trauma. Pain has continued to increase throughout the day and has been unresponsive to oral analgesia. He is alert and oriented. He has marked swelling about his leg. He has decreased sensation on the dorsum of his foot. Peripheral pulses remain palpable. Pain is exacerbated anteriorly by passive plantar flexion of his great toe. The leg is firm and incompressible. He has WBC of 1600/mm³, hemoglobin of 7.3 g/dL, and platelet count of 8000/mm³. Radiographs are unremarkable. What is the most likely diagnosis and what additional tests are necessary prior to any definitive intervention?
 - (a) Pathologic fracture. CT scan should be obtained to better clarify the fracture.
 - (b) Compartment syndrome. No additional studies are needed.
 - (c) Abscess. MRI should be performed to both confirm diagnosis and assess for additional pathologic findings.
 - (d) Compartment syndrome. A manometer (i.e., Stryker needle) should be used in all four compartments of the leg to confirm diagnosis prior to surgery.
- 97. DRESS/DIHS (drug reaction with eosinophilia and systemic symptoms/drug-induced hypersensitivity syndrome) is most commonly characterized by which of the following?
 - (a) Skin pain, skin blistering, targetoid lesions, fever
 - (b) Maculopapular rash, facial edema, fever, eosinophilia
 - (c) Urticaria and angioedema
 - (d) Pustules, fever, neutrophilia

- 98. Which of these statements regarding immune-related cutaneous adverse events (ircAEs) to immunotherapy is false?
 - (a) The most common manifestations are pruritus and maculopapular rash.
 - (b) They can be distinguished from cutaneous adverse drug reactions to regular medications by skin biopsy.
 - (c) They may be associated with prolonged progression-free survival and overall survival.
 - (d) They can occur at any time during or even after treatment.
- 99. Which of the following skin morphologies can reflect cutaneous blood vessel lumen occlusion (i.e., thrombosis/embolic)?
 - (a) Macular purpura
 - (b) Palpable purpura
 - (c) Ecchymoses
 - (d) Retiform purpura
- 100. Which of the following is *not* a criterion of SCORTEN severity illness score for predicting mortality in SJS/ TEN?
 - (a) Number of mucosal sites involved
 - (b) Heart rate >120
 - (c) Age >40
 - (d) Initial percentage of epidermal detachment >10%
- 101. A 4-year-old girl undergoing chemotherapy for acute lymphoblastic leukemia and is known to be neutropenic presents with new onset abdominal pain and tenderness in the right lower quadrant on physical examination. A rare but potentially fatal complication of pretreatment is diagnosed on ultrasound. The findings on ultrasound that would support this include:
 - (a) A "target sign" found in the right lower quadrant
 - (b) A blind ended bowel loop with a noncompressible hyperemic bowel wall
 - (c) A cystic structure in the periumbilical region
 - (d) A thickened bowel wall located in the right lower quadrant
 - (e) A "whirlpool sign" in the right lower quadrant
- 102. A patient who recently started chemotherapy for Burkitt's lymphoma with high tumor burden is being monitored for complications of therapy. Laboratory testing of the blood would be expected to show:
 - (a) Low potassium
 - (b) Low calcium
 - (c) Low phosphorus
 - (d) Low urea nitrogen
 - (e) Low sodium
- 103. As pediatric oncologic patient survival rates increase due to advances in medications and treatment protocols, delayed mortality is most notably due to:

- (a) Sepsis
- (b) Suicide
- (c) Arrhythmias
- (d) Interstitial lung disease
- 104. Pediatric cancers present in many ways, typically with vague complaints. Which of the following is not a typical presentation seen?
 - (a) Fatigue
 - (b) Weight gain
 - (c) Bone pain
 - (d) Fever
 - (e) Pallor
- 105. A 69-year-old man is in the ED with worsening shortness of breath. His wife also states that he appears more confused lately and has not had an appetite in the last couple of days. His white blood sound count is 120,000/mm³ with 93% blasts. He has a history of acute myeloid leukemia with plans to start treatment in the next few days. On physical examination, the patient is only oriented to self, oxygen saturation is 92% on room air, heart rate is 113, but afebrile. Chest radiographs indicate bilateral infiltrates, but no focal lesions or consolidations, pneumothorax, or pleural effusion. Oncologic was consulted and will see the patient. Meanwhile, what is the initial management in the ED? (a) Antibiotics
 - (b) Hydroxyurea
 - (c) Isotonic intravenous fluids
 - (d) Leukapheresis
- 106. A 75-year-old woman with a history of ALL is brought to the ED by her family for "acting funny." Her daughter reports that her mother has seemed more confused over the past few days, and now is not using her right hand to feed herself. On physical examination, vital signs are all within normal limits. The patient is disoriented and has right-sided weakness. CBC is notable for WBC 60,000/mm³, platelet count 6000/mm³, and hemoglobin 6.5 g/dL. Which of the following is the best initial diagnostic test to obtain?
 - (a) CT head without contrast
 - (b) MRI brain
 - (c) Lumbar puncture
 - (d) EEG
- 107. A 39-year-old man without major past medical history is sent in from the local urgent care for abnormal laboratory work. He reports feeling run down and fatigued with a decreased appetite and nonproductive cough for the past few weeks, and thought he maybe had the flu. His vital signs are notable for a heart rate of 128 and a respiratory rate of 30 but are otherwise within normal limits. He appears pale, dry, and tachypneic with mild retractions, but denies shortness of breath. His paper-

work from urgent care shows a WBC count of 205,000/ mm³, hemoglobin of 7 g/dL, platelets of 18,000/mm³, creatinine of 3 mg/dL, a bicarbonate of 12 mmEq/L, a lactate of 4 mmol/L, troponin of 1.9 ng/mL, and a venous blood gas with a pH of 7.25 and a PCO₂ of 27 cmH₂O. His EKG shows sinus tachycardia with lateral ST segment depressions. You start IV fluids and send a page to the on-call oncologist immediately. What is the best way to manage this patient while waiting for the oncologist to return your page?

- (a) Give aspirin 324 mg and start a heparin drip
- (b) Transfuse packed red blood cells to a goal hemoglobin of 8–10 mg/dL
- (c) Administer empiric antibiotics
- (d) Intubate
- 108. A 56-year-old man with history of hypertension and chronic back pain presents with a triage complaint of "back pain." He is moaning a lot and not providing a lot of answers to your questions. His family reports that they knew something was wrong when he did not go to his methadone clinic today. Vital signs: temperature 37.3 °C (99.1 °F), heart rate 104 beats per minute, blood pressure 156/93, respiratory rate 27 breaths per minute, oxygen saturation 94% on room air. You obtain lab work that demonstrates a WBC count of 87,000/ mm³, hemoglobin of 5.9 g/dL, platelets of 9000/mm³, potassium of 5.6 mmol/L, CO2 of 15 mmol/L, a creatinine of 2.7 mg/dL, a calcium of 7.6 mg/dL, and a phosphorus of 6 mg/dL. His chest X-ray is clear, and his lumbar spine X-ray is consistent with degenerative joint disease. You consult the on-call oncologist who asks for a manual differential, uric acid, LDH, coagulation studies, initiation of intravenous fluids, and admission to the hospitalist. What is the appropriate next step in this patient's care?
 - (a) Initiate sodium bicarbonate infusion
 - (b) Obtain a noncontrasted CT scan of the head
 - (c) Transfuse 2 units of packed red blood cells and 1 pack of platelets
 - (d) Administer IV calcium
- 109. For patient with cancer and acquired factor VIII inhibitors, eradication of the tumor is crucial to controlling the inhibitor.
 - (a) True
 - (b) False
- 110. Which one of these antineoplastic therapies has been associated with low antithrombin levels?
 - (a) Bevacizumab
 - (b) Hydroxyurea
 - (c) L-Asparaginase
 - (d) Thalidomide

- 111. Which mutation is associated with thrombosis in myeloproliferative neoplasms?
 - (a) BCR-ABL
 - (b) CALR
 - (c) JAK2
 - (d) MPL
- 112. A rare cause of bleeding in patients with myeloproliferative neoplasms is acquired deficiency of factor:
 - (a) V
 - (b) VII
 - (c) X
 - (d) XI
- 113. A 19-year-old woman with a history of sickle cell disease presents with pain in her legs. She says it is her typical pain crisis and denies shortness of breath, chest pain, or nausea. Laboratory testing reveals the following: WBC count, 10,000/mcL; Hgb, 4 g/dL; and reticulocyte count, 0.5%. Electrolytes are normal. She responds well to pain medication. Which of the following treatments is helpful for her condition?
 - (a) Antibiotic therapy
 - (b) Exchange transfusion
 - (c) Oxygen administration
 - (d) Transfusion of packed red blood cells
- 114. A 24-year-old man presents to the ED after 1 week of worsening left shoulder pain. He has had pain over the past 2 months intermittently. He denies any fever, trauma, swelling, or rash. He has a history of sickle cell disease. His vital signs are normal. You order an X-ray of the left shoulder. What finding is likely to be demonstrated on the X-ray?
 - (a) Avascular necrosis
 - (b) Fracture of the humerus
 - (c) Osteomyelitis
 - (d) Tumor in the lung
- 115. A 19-year-old male presents to the ED 1 week after his last visit complaining of pain in his legs that is typical of his symptoms on prior visits. When should laboratory tests including a CBC, reticulocyte count, LFTs, bilirubin, LDH, and electrolytes be considered?
 - (a) The patient is being admitted.
 - (b) You suspect another diagnosis.
 - (c) The patient appears toxic.
 - (d) All of the above.
- 116. A 38-year-old woman with a known history of sickle cell disease comes into the ED 1 week after her last uneventful pain crisis. She complains of 3 days of non-productive cough and gradual onset severe central chest pain radiating to both shoulders. She admits to being short of breath on exertion. Her vitals show a temperature of 38.2 °C, heart rate of 108/min, blood

pressure of 126/78 mmHg, and an oxygen saturation of 92% on room air. She is short of breath on examination and her lung sounds are clear. She has a normal cardio-vascular exam. A chest X-ray is ordered. Which of the following chest radiograph findings is likely to be present?

- (a) Multilobar infiltrates
- (b) Pneumothorax
- (c) No acute pathology
- (d) Rib fracture
- 117. A 19-year-old man presents to the ED with acute change in mental status. Over the past 3 weeks, he has been experiencing progressive fatigue, easy bruising, and new bleeding from his gums. In the days prior to admission, he developed headache in the left periorbital area and recently complained to his family of double vision. His physical examination is concerning for an obtunded patient who is responsive to questions but incoherent. He has esotropia of the left eye. Initial laboratory results demonstrate anemia, thrombocytopenia, and an elevated white count that is predominantly lymphocytic. His other leukocyte lineages are suppressed, including neutrophils, with an absolute neutrophil count of 400 cells per uL. CT head demonstrates acute inflammation of the left nasal sinus and associated periorbital fat stranding with a radiologist impression of acute invasive fungal sinusitis. Which of the following is the best next step?
 - (a) Order STAT MRI
 - (b) Start liposomal amphotericin B
 - (c) Start micafungin
 - (d) Order NP swab for culture
- 118. A 62-year-old woman with metastatic cholangiocarcinoma currently receiving gemcitabine plus cisplatin presents to the ED for fever and abdominal pain. Her second cycle of chemotherapy was given in the outpatient cancer infusion center via her port approximately 1 week ago. On initial evaluation, she is febrile but alert and stable. She notes severe right lower abdominal pain which comes in waves. She states for several days she has noted diarrhea; however, today it became maroon in color before stopping. Initial laboratory work demonstrates a hemoglobin of Hgb 9.2 mg/dL, a platelet count of 112,000 per μ L, and an ANC of 250 cells per μ L. She is started on empiric cefepime and metronidazole. Which of the following is the best next step?
 - (a) CT scan of the abdomen
 - (b) Add IV vancomycin to the empiric antibiotics
 - (c) Give colony-stimulating factor for neutropenia
 - (d) Send stool C difficile toxin assay

- 119. A 65-year-old woman with metastatic breast cancer receiving chemotherapy presents to the ED with fever, chills, and night sweats. She has a tunneled internal jugular central venous catheter in place through which she has been receiving chemotherapy as well as intravenous fluids and total parenteral nutrition per home nursing due to dehydration and poor oral intake. Vital signs are notable for an oral temperature of 38.4 °C, BP 100/60 mmHg, and HR 72 bpm. She appears chronically ill but is in no acute distress. The insertion site for the catheter is notable for mild erythema but no induration or drainage. There is no tenderness to palpation along the subcutaneous tract of the catheter. Which of the following is the best next step?
 - (a) Start empiric IV vancomycin and cefepime
 - (b) Draw 2 sets of blood cultures from the central venous catheter
 - (c) Obtain 2 sets of blood cultures from separate peripheral venous sites
 - (d) Coordinate immediate removal of the central venous catheter
- 120. A 54-year-old man with a history of hematopoietic stem cell transplant (HSCT) 6 months ago presents with worsening cough and dyspnea on exertion over the past 3 weeks. His HSCT has been complicated by graft-versus-host disease, requiring chronic immuno-suppression including high-dose corticosteroids. The patient is currently taking valganciclovir and flucon-azole; he was also supposed to be taking trimethoprim-sulfamethoxazole, but this was held approximately 2 months back due to acute kidney injury. In triage, he is notably hypoxic with the pulse oximeter reading 86% on room air. A chest radiographic demonstrates diffuse ground glass opacities. With which of the following opportunistic infections is this patient's clinical presentation most consistent?
 - (a) CMV pneumonia
 - (b) HSV pneumonia
 - (c) Pseudomonas aeruginosa pneumonia
 - (d) Pneumocystis jirovecii pneumonia
- 121. A 16-year-old male with leukemia presents to the ED with a temperature of 38.5 °C and a mild, dry cough. His vital signs are otherwise within normal limits, and his laboratory results are unremarkable. His fever resolves without treatment and he feels otherwise well. In addition to initiating antibiotic therapy, what is the next step in this patient's care?
 - (a) Apply the MASCC score to risk stratify the patient
 - (b) Apply the CISNE score to risk stratify the patient

- (c) Place the patient in observation status
- (d) Admit the patient for ongoing antibiotics and close monitoring
- 122. A 43-year-old female with a history of osteosarcoma presents to the ED with a temperature of 39 °C. It has been approximately 7 days since her last chemotherapy and she is found to be neutropenic. She reports using a mask and gloves at all times and is wondering how she developed this infection. What is the most likely source?
 - (a) She likely has a noninfectious cause such as drug or tumor fever.
 - (b) She probably contracted her infection from her kids.
 - (c) She most likely developed the infection from endogenous flora.
 - (d) She likely became infected during her last infusion center visit.
- 123. The following are considered safe discharge criteria for patients with febrile neutropenia except:
 - (a) Residence ≤4 hours or ≤100 miles (160 km) from clinic or hospital
 - (b) Access to a telephone and transportation 24 h/d
 - (c) Family member or caregiver at home 24 h/d
 - (d) No history of noncompliance with treatment protocols
- 124. A patient with leukemia has been diagnosed with lowrisk febrile neutropenia is cleared for discharge, but he is very concerned that he will not be receiving intravenous antibiotics. What does the evidence say regarding the use of oral versus intravenous antibiotics in febrile neutropenia?
 - (a) Oral antibiotics are inferior to intravenous antibiotics in all cases.
 - (b) Oral antibiotics and intravenous antibiotics are equally effective.
 - (c) Oral antibiotics are noninferior to intravenous antibiotics in cases of low-risk febrile neutropenia.
 - (d) Oral antibiotics are never indicated in patients with hematologic malignancies.
- 125. A 70-year-old man with acute myeloid leukemia presents to the ED after suddenly developing fever, diffuse abdominal pain, and diarrhea. You are very concerned for neutropenic enterocolitis and consult the hematology/oncologic and surgical teams. Based on the above presentation, your diagnostic approach should include:
 - (a) Immediate surgical consultation and transfer to the operating room for bowel resection.
 - (b) Evaluation for GVHD and *C. diff* as well as CT imaging to help characterize the diagnosis.
 - (c) Bedside ultrasound to make a definitive diagnosis followed by administration of antibiotics and admission.

- (d) Immediate administration of broad-spectrum antibiotics and ICU admission, CT imaging is unnecessary at this time.
- 126. A 70-year-old man with acute myeloid leukemia presents to the ED after suddenly developing fever, diffuse abdominal pain, and diarrhea. On physical examination, he has right lower quadrant abdominal pain with associated guarding and bowel sounds are present. Laboratory evaluation reveals a white blood cell count of 1200 cells/µL (ANC = 100/dl), a hematocrit of 23%, and 40,000 platelets/µL. Computed tomography (CT) scan of the abdomen and pelvis shows thickening of the colonic wall with diffuse pericolic edema and pneumatosis. What additional information will help guide a decision regarding surgery?
 - (a) Extent of bowel wall thickening
 - (b) Persistent GI bleeding
 - (c) Involvement of the cecum
 - (d) Chemotherapeutic regimen
- 127. Risk factors for neutropenic enterocolitis in the above patient include all of the following, except:
 - (a) Specific chemotherapeutic agent
 - (b) Timing from last chemotherapy
 - (c) Route of administration of chemotherapy
 - (d) Tumor type (i.e., solid vs. hematogenous)
- 128. Blood cultures and stool studies for *Clostridium difficile* and other enteropathogenic organism are pending for the above patient. Besides broad-spectrum antibiotics and close hemodynamic monitoring, what additional therapies might the patient benefit from?
 - (a) Recombinant granulocyte colony-stimulating factor (G-CSF)
 - (b) Corticosteroid
 - (c) Rasburicase
 - (d) Recombinant factor VIIa
- 129. What are the three domains of biopsychosocial model?
 - (a) Hypothalamus-pituitary-adrenal axis
 - (b) Predisposing, precipitating, and perpetuating factors
 - (c) Childhood trauma, anger, and impulsive behavior
 - (d) Environmental, financial, and psychosocial factors
- 130. Cancer patients have a higher risk for suicide as compared to general population, because:
 - (a) Cancer is common in older frail people.
 - (b) They have financial crisis due to cancer treatment.
 - (c) They lose their job due to the disease.
 - (d) In addition to general risk factors of suicide, they have the burden of adverse effects of cancer treatment.
- 131. How do you differentiate depression from side effects of cancer treatment?
 - (a) Hopelessness and loss of interest in previously enjoyable activities

- (b) Poor sleep
- (c) Fatigue
- (d) Weight loss
- 132. What would you do as the first step when discharging a low-risk suicidal patient?
 - (a) Give a prescription of antidepressant
 - (b) Arrange a follow-up with a therapist
 - (c) Making sure patient has no access to weapons
 - (d) Giving him a suicide hot line number
- 133. All the following information about delirium in the cancer patient present are correct, except:
 - (a) Delirium is different from altered mental status.
 - (b) May be interpreted as worsening pain.
 - (c) Most delirium cases in ED are of the agitated (hyperactive) type.
 - (d) It is frequently missed by ED providers.
- 134. What are the most common causes of delirium in the cancer patient presenting to ED:
 - (a) Brain metastasis
 - (b) Liver failure
 - (c) Renal failure
 - (d) Side effects of medications
 - (e) Paraneoplastic syndrome
- 135. What is the most important step in the management of the cancer patient with delirium in the ED?
 - (a) Give benzodiazepine intravenously.
 - (b) Intubate the patient.
 - (c) Give any of the atypical antipsychotic medications such as olanzapine.
 - (d) Assure safety of the patient.
 - (e) Give haloperidol.
- 136. Which of the following statements about palliative sedation is correct?
 - (a) Haloperidol is commonly used for this procedure.
 - (b) Refractory delirium is the most common indication for this procedure among cancer patients.
 - (c) When stated, it should be continued until the patient die.
 - (d) May be started if the patient (or family) wishes to shorten life.
- 137. Transalar herniation can lead to infarction in which territories?
 - (a) Anterior cerebral artery (ACA)
 - (b) Middle cerebral artery (MCA)
 - (c) Posterior cerebral artery (PCA)
 - (d) A and B
 - (e) A, B, and C
- 138. What is the most common cause of massive hemoptysis worldwide?
 - (a) Bronchogenic carcinoma
 - (b) Trauma
 - (c) Postoperative complication

- (d) Pulmonary fibrosis
- (e) Pulmonary tuberculosis (TB)
- 139. What is the approximate attenuation in Hounsfield units (HU) of acute blood products on CT?
 - (a) -1000
 - (b) -100
 - (c) 0
 - (d) 40
 - (e) 1000
- 140. At least what percent of bone mineral density must be lost in order for radiographs to detect lytic lesions?
 - (a) 10%
 - (b) 30%
 - (c) 50%
 - (d) 70%
 - (e) 90%
- 141. A 48-year-old female with history of ovarian cancer presents with lethargy. Her husband who accompanies her notes that she was scheduled for a chemotherapy infusion today in clinic but she was "too sick to go through with it." Her vital signs: temperature 101.2 °F (38.4 °C), pulse 112 bpm, respirations 24 bpm, blood pressure 88/50, and pulse oximetry 98% on room air. She looks chronically ill, was lethargic but awake and spontaneously breathing, and was unable to provide much history. The abdomen appears distended and is generally tender to palpation. An IV is placed, a Foley is inserted with clear urine returned, and fingerstick glucose is 120. You perform a bedside ultrasound that shows the following finding on the left side only (Fig. 1). In addition to obtaining blood cultures, administering broad-spectrum antibiotics and 30 ml/kg of crystalloid fluids, what action should you take?



Fig. 1 Ultrasound findings

- (a) Initiate dobutamine, and hold the fluids with concern for cardiogenic shock.
- (b) Consult a urologist to request an emergent nephrostomy.
- (c) Consult a general surgeon to request an emergent laparotomy.
- (d) Consult a vascular surgeon for catheter-directed thrombolysis.
- (e) Place an NG tube, make the patient NPO, and obtain a CT abdomen-pelvis to evaluate for a bowel obstruction.
- 142. A 22-year-old male with a history of metastatic testicular cancer presents with chest pain that began approximately 3 hours ago when he woke up. It is sharp, worse with lying flat, and partially relieved with sitting up, associated with shortness of breath and dizziness. Vital signs: temperature 98.7 °F (37.0 °C), pulse 104 bpm, respirations 20 bpm, blood pressure 103/76 by automatic cuff, and oxygen saturation 97% on room air. ECG shows sinus rhythm, rate of 104, normal axis, mild (<1 mm) ST segment elevation in all the precordial leads without depressions, and no TW inversions. The patient is thin, awake, alert, and oriented, with no obvious abnormality on cardiopulmonary examination. Bedside echocardiography reveals the following finding (Fig. 2). Which of the following management options should be considered?
 - (a) Catheter-directed thrombolysis
 - (b) Cardiac catheterization
 - (c) Placement of a left-sided pigtail catheter for tension pneumothorax
 - (d) Left thoracentesis
 - (e) Pericardiocentesis
- 143. A 39-year-old female with history of breast cancer who is currently undergoing chemotherapy presents to the ED with leg swelling for 1 week. Her vital signs were only



Fig. 2 Ultrasound findings

remarkable for sinus tachycardia of 108 bpm. Upon examination, you find 3+ edema on the right lower extremity up to the mid-thigh. There is significant right calf and thigh tenderness with overlying erythema. Which of the following would confirm the suspected diagnosis?

- (a) A noncompressible common femoral artery
- (b) A noncompressible common femoral vein with an echogenicity in the lumen
- (c) A compressible common femoral vein with the presence of luminal flow
- (d) A compressible popliteal artery with luminal echogenicity
- 144. A 50-year-old female with breast cancer presents with shortness of breath and dyspnea on exertion. She is currently receiving doxorubicin as a chemotherapeutic agent. She has not previously had symptoms like this. Examination is remarkable for hypoxia to 90% on room air, jugular venous distension (JVD) and symmetric bilateral lower extremity edema, and obvious orthopnea. Which of these findings would confirm the suspected diagnosis?
 - (a) Diffuse B lines and bilateral pleural effusions on POCUS lung examination and a decreased left ventricular ejection fraction on cardiac ultrasound.
 - (b) Diffuse A lines on POCUS lung examination and a normal left ventricular ejection fraction on cardiac ultrasound.
 - (c) Diffuse A lines on POCUS lung examination and a hyperdynamic left ventricular ejection fraction on cardiac ultrasound.
 - (d) Significant free fluid in the abdomen on RUSH examination and a dilated IVC.
 - (e) Focal B lines and unilateral pleural effusion on POCUS examination of the right lung and a normal left ventricular ejection fraction on cardiac ultrasound.
- 145. A 50-year-old African-American female with breast cancer is receiving paclitaxel as part of her chemotherapy regimen. On a follow-up visit, she complains of symmetrical paresthesias in her legs which began in her toes a few weeks ago. She has tried over the counter pain medications with no relief. Her past medical history is significant for diabetes for which she takes metformin. She takes no other medications at this time. Which is the best evidenced-based treatment option?
 - (a) Gabapentin 300 mg po every 8 hours
 - (b) Duloxetine 30 mg po daily for 1 week then increase to 60 mg po daily
 - (c) Baclofen 10 mg, amitriptyline 40 mg, and ketamine 20 mg in pluronic gel apply to affected area three times daily
 - (d) Acupuncture

- 146. A 42-year-old male presents to the ED following extravasation of vincristine. The needle has been removed prior to presentation to the ED and the area of extravasation is red, swollen, and painful to touch. At this time, there is no ulceration and the affected area appears to not be progressing. Which of the following is the appropriate management for this patient?
 - (a) Administer hyaluronidase to areas of extravasation
 - (b) Do not elevate arm
 - (c) Apply cold compress
 - (d) Consult plastic surgery to evaluate the area
- 147. What is the most important agent for management of a patient with anaphylaxis secondary to cisplatin?
 - (a) Epinephrine IM 0.5 mg into the deltoid muscle
 - (b) Diphenhydramine 50 mg IV
 - (c) Epinephrine IM 0.01 mg/kg IM into lateral aspect of thigh (max dose 0.5 mg)
 - (d) Hydrocortisone 50 mg IV
- 148. A 34-year-old female who received her first cycle of fluorouracil, methotrexate, and cyclophosphamide as part of her chemotherapy regimen for breast cancer. The following day she presents to the ED with complaints of severe fatigue and confusion. Her husband states she was in her normal state of health until she woke up this morning with acute confusion. Her vital signs are normal with the exception of a slighted elevated blood pressure at 185/93 mmHg. Her home medications include sertraline 50 mg orally daily for depression and tramadol 50 mg orally as needed for pain. She has no other significant past medical history. A noncontrast CT scan of the brain in the ED is unremarkable. A recent MRI from 1 week ago showed no brain metastasis. What is the most likely cause of the patient's confusion?
 - (a) Undiagnosed brain metastasis
 - (b) Severe SIADH resulting in hyponatremia
 - (c) Infection
 - (d) Head trauma
- 149. Radiation treatment to the heart is associated with injury to:
 - (a) Valves
 - (b) Myocardium
 - (c) Coronary arteries
 - (d) All of the above
- 150. Radiation injury to the heart is influenced by:
 - (a) Radiation dose
 - (b) Volume in the therapy field
 - (c) Daily fraction size
 - (d) All of the above
- 151. Radiation injury to the lung can occur outside the radiation treatment field.
 - (a) True
 - (b) False

- 152. Radiation injury to the kidney generally spares the glomeruli.
 - (a) True
 - (b) False
- 153. With medical support, the LD 50/30 for total body radiation exposure is generally thought to be 5 cGy.
 - (a) True
 - (b) False
- 154. Lymphocytes may serve as a surrogate biomarker for radiation dose during an acute exposure incident.
 - (a) True
 - (b) False
- 155. Bone marrow transplant is a proven treatment for acute radiation exposure.
 - (a) True
 - (b) False
- 156. At least what percentage of pediatric patients undergoing hematopoietic stem cell transplant will experience oral and/or gastrointestinal mucositis?
 - (a) 25%
 - (b) 45%
 - (c) 65%
 - (d) 85%
 - (e) 95%
- 157. Which of the following costs has *most contributed* to the economic burden related to oral and gastrointestinal mucositis?
 - (a) Administration of antimucositis agents
 - (b) Invasive procedures implemented for diagnosis of mucositis
 - (c) Extended inpatient hospitalizations related to potential sequelae of mucositis
 - (d) Additional time spent by physicians and clinical staff managing patient symptoms of mucositis
 - (e) Enteral feeding tube intervention
- 158. Which of the following agents is appropriate for the prophylaxis of oral mucositis in a patient undergoing chemoradiation therapy for cancer of the head and neck?
 - (a) Benzydamine mouthwash
 - (b) Oral antibiotics
 - (c) Hyperbaric oxygen therapy
 - (d) Prebiotic supplementation
 - (e) Fat-modified diet
- 159. Which of the following statements is true regarding mucositis?
 - (a) Patient-reported duration and severity of mucositis are typically shorter and less severe compared to that reported by physicians.
 - (b) Endoscopic evaluation is preferred for the diagnosis of gastrointestinal mucositis.
 - (c) Early intervention with intravenous fluid hydration and/or enteral feeding tube placement is critical in

patients experiencing rapid, continuous weight loss and can reduce the risk of hospitalization or treatment break.

- (d) Geriatric patients are at low risk of developing oral and gastrointestinal mucositis.
- (e) Bacterial flora present in the oral and gastrointestinal mucosae have no impact on the pathogenesis of mucositis.
- 160. A 54-year-old male with a history of an allogeneic HCT 6 months ago for acute myeloid leukemia presents to the ED with profound melena. Vital signs showed a BP of 90/60, HR of 120, RR 18, and T 36.8 °C. CBC showed Hgb of 6 g/dL, Hct of 19%, WBC of 9.3/uL, and platelets of 150,000/uL. You decide to transfuse packed red blood cells.
 - (a) Washed RBC
 - (b) Genotypically matched RBC
 - (c) Leukoreduced RBC
 - (d) Irradiated RBC
 - $(e) \ Both \ C \ and \ D$
- 161. A 64-year-old man with AL amyloidosis who is undergoing autologous stem cell mobilization with highdose G-CSF presents to the ED with lightheadedness and severe abdominal pain. Which of the following complications of high-dose G-CSF do you suspect?
 - (a) Pancreatitis
 - (b) Esophageal rupture
 - (c) Splenic rupture
 - (d) Mesenteric ischemia
- 162. A 52-year-old man with B-cell acute lymphoblastic leukemia receives nonmyeloablative conditioning followed by growth factor-mobilized blood cell transplantation from an HLA-matched unrelated donor. GVHD prophylaxis consists of cyclosporine, sirolimus, and mycophenolate mofetil. Twenty-one days after transplant, the patient develops new anorexia, early satiety, nausea, and low-volume diarrhea (approximately 300 mL/day). Despite scheduled antiemetics and diet adjustment, his GI symptoms persist. The patient has lost 5.0 kg over 1 week. The patient and donor were both CMV seropositive. Which of the following represents the best approach to treating this patient?
 - (a) Discontinue mycophenolate mofetil as this is the most likely cause of GI symptoms.
 - (b) Treat empirically with ganciclovir for presumed CMV enteritis.
 - (c) Refer to gastroenterology for consideration of endoscopic studies to evaluate for acute GVHD and test for viral infection.
 - (d) Treat empirically with prednisone at 2.0 mg/kg/ day.

- 163. A 22-year-old patient with peripheral T-cell lymphoma presents to the ED with language incoherence noticed by his wife and gait change of 12-hour duration. The patient feels well and states he feels it is normal since he recently underwent CD19-chimeric antigen receptor (CAR)-T-cell therapy.
 - (a) Calculate the immune effector cell-associated encephalopathy (ICE) score since the symptoms can be a delayed presentation of immune effector cell-associated neurotoxicity syndrome (ICANS).
 - (b) Call the stroke team.
 - (c) Refer the patient to neurology clinic.
 - (d) Discharge the patient without further evaluation since the patient feels normal.
- 164. A 65-year-old patient with advanced melanoma has received 3 cycles of the combination of nivolumab and ipilimumab, and now presents to the ED with chest pain. He has no risk factors or family history of coronary artery disease. His electrocardiogram shows no ST segment changes suggestive of myocardial ischemia or pericarditis. What is the best next step in management?(a) Discharge the patient if normal initial troponin level.
 - (b) Obtain D-dimer and 3 sets of cardiac enzymes. Initiate aspirin and anticoagulation (if no brain metastasis). Rule out myocarditis from immune checkpoint inhibitors.
 - (c) Start the patient on dexamethasone.
 - (d) None of the above.
- 165. A patient with acute myeloid leukemia presents to the ED with fever, hypotension, renal failure, and transaminitis. What is the appropriate management if the patient presents a card stating he is on oral therapy with enasidenib, which was started 3 weeks ago?
 - (a) Draw cultures, initiate intravenous fluids 30 mL/ kg, and start broad-spectrum antibiotics.
 - (b) Admit to the intensive care unit.
 - (c) Consider the diagnosis of differentiation syndrome and initiate dexamethasone therapy after discussion with the oncologist.
 - (d) All of the above.
- 166. A patient with acute lymphoblastic leukemia presents to the ED for sudden onset fever, chills, and shortness of breath about 15 minutes upon reaching home after receiving blinatumomab infusion in the ambulatory treatment center. What is the best next step in management?
 - (a) Assess the patient's airway for vital signs and respiratory status and administer oxygen support.
 - (b) Evaluate for acute infection, probable neutropenic fever, and empirically start broad-spectrum antibiotics.

- (c) Treat with corticosteroid, antihistamines, and inhaled beta-agonists for a delayed infusion reaction syndrome or cytokine release syndrome.
- (d) All of the above.
- 167. What is one of the key indicators of poor quality care with regard to ED visits at the EOL?
 - (a) More than 2 visits in the last 6 months of life
 - (b) Avoidable visits in the last year of life
 - (c) More than 1 ED visit in the last 30 days of life
 - (d) Less than 1 ED visit in the last 6 months of life
- 168. What is the benchmark population value rate that regions should strive to be below for patients with cancer accessing ED care?
 - (a) 50%
 - (b) 80%
 - (c) 10%
 - (d) 30%
- 169. Which of the following is not among the 3 most common reasons for ED visits for patients with cancer at end of life?
 - (a) Fever
 - (b) Pain
 - (c) Shortness of breath
 - (d) Nausea
- 170. Research has shown that receipt of palliative care services is associated with various outcomes. Which of the following is not one of them?
 - (a) Better symptom control
 - (b) Improved knowledge of expectations
 - (c) Shorter survival
 - (d) Decreased chemotherapy use
- 171. Patients can develop tolerance to which adverse effect of opioids?
 - (a) Respiratory depression
 - (b) Constipation
 - (c) Pruritus
 - (d) All of the above
- 172. Which of the following statements is true about neuropathic pain?
 - (a) Peritoneal carcinomatosis generally causes neuropathic pain.
 - (b) Anticonvulsants require close titration of dosing when treating neuropathic pain.
 - (c) Topical lidocaine patches should be worn continuously for maximum treatment of neuropathic pain.
 - (d) Corticosteroids are contraindicated in neuropathic pain.
- 173. Which opioid does not follow first-order kinetics?
 - (a) Morphine
 - (b) Dilaudid
 - (c) Methadone
 - (d) Fentanyl

- 174. A 40-year-old woman presents to the ED in severe pain that she rates as 8/10. Her only past medical history is breast cancer. Twenty minutes after receiving a dose of IV morphine, she has no adverse side effects of the morphine but her pain remains 8/10. What is the best next step for her pain control?
 - (a) Repeat the same dose of IV morphine at this time.
 - (b) Re-assess the patient in 30 minutes because the first dose has not had sufficient time to work.
 - (c) Increase the initial dose by 50–100% and administer a second IV dose at this time.
 - (d) Switch to intramuscular morphine for more reliable absorption.
- 175. A 61-year-old man with a history of metastatic prostate cancer currently receiving radiation presents to the ED complaining of constipation. He is taking morphine daily for pain control but does not take medications to prevent constipation. After ruling out stool impaction and bowel obstruction, you discuss the case with the patient's oncologist and determine the patient is safe for discharge. You plan to send him home with a prescription to ease his constipation? Which of the following is true regarding medications for opioid-induced constipation prophylaxis?
 - (a) Methylnaltrexone is considered the best first-line agent for opioid-induced constipation prophylaxis.
 - (b) Stimulants and osmotic agents are often used together.
 - (c) Stimulants act by increasing water content in the large bowel.
 - (d) Polyethylene glycol is considered a bulk-forming agent.
- 176. The risk factors for a substance abuse problem developing in a person being treated for cancer pain with opioids include all of the following except:
 - (a) Older age
 - (b) Male gender
 - (c) Personal psychiatric history
 - (d) Family history of substance abuse
- 177. Pseudo addiction refers to a syndrome in which a person in poorly controlled pain develops aberrant behaviors that are uncharacteristic of them under normal circumstances and are expected to resolve when adequate pain control is provided. Which of the following are true of pseudo addiction:
 - (a) Pseudo addiction is mutually exclusive from actual loss of control of one's medications and the development of drug abuse.
 - (b) The behaviors always resolve with the provision of improved analgesia alone.
 - (c) It is best thought of as similar to secondary alcoholism.

- (d) The original concept came from a large clinical trial.
- 178. In a study examining documentation of alcohol problems in 100 alcoholics with advanced illness, how many had their alcohol history noted in their medical record?
 - (a) 80%
 - (b) 60%
 - (c) 55%
 - (d) 30%
- 179. Disulfiram (Antabuse) has been shown to work in some highly selected patients with alcoholism. Which of the following characteristics does not predict the likelihood of a response to Antabuse?
 - (a) Older than 40 years of age
 - (b) Shorter drinking histories
 - (c) Socially stable
 - (d) Highly motivated
- 180. A 70-year-old patient with advanced lung cancer, enrolled in hospice care, presents to the ED with severe shortness of breath. The patient's goals are to focus on comfort and quality of life and to stay at home, but his caregiver panicked and called 911 when his breathing worsened. On physical examination, the patient reports severe dyspnea and appears very uncomfortable with a RR of 35 with accessory muscle use. The most important next step is to:
 - (a) Intubate the patient
 - (b) Place the patient on BIPAP
 - (c) Administer a test dose of morphine 2 mg IV and reassess for symptomatic improvement
 - (d) Get a STAT chest X-ray
- 181. In patients with advanced cancer presenting in respiratory distress, noninvasive ventilation
 - (a) Should always be used as a first step
 - (b) May be beneficial to some and can be utilized if consistent with the patient's goals of care
 - (c) Is rarely helpful and should not be considered
 - (d) Leads to recovery with discharge to home in nearly all patients
- 182. Diagnoses that commonly lead to dyspnea in patients with advanced cancer include:
 - (a) Pleural effusions
 - (b) Anemia
 - (c) Tumor burden in lungs
 - (d) All of the above
- 183. A 60-year-old patient with advanced pancreatic cancer is brought in by EMS after being found down in her home. She was with her friend at the time of the event. EMS arrived and found the patient in cardiac arrest, with an initial rhythm of PEA. ACLS protocol was initiated, the patient was intubated, and eventually ROSC

was achieved. The patient's spouse has now arrived in the ED and says that the patient was DNR/DNI and did not want to be kept alive on machines at this stage of her illness. The most appropriate next step would be to:

- (a) Walk into the room and remove the ETT.
- (b) Counsel the patient's spouse that extubation is not allowed.
- (c) Provide emotional support to the patient's spouse, counsel him on the patient's expected prognosis off the ventilator, and then start gathering the appropriate medications for extubation.
- (d) Explain that extubations are only possible in the ICU.
- 184. Which of the following statements is true regarding outcomes treatments in cancer patients?
 - (a) Most patients with stage IV lung cancer understand that chemotherapy will cure their cancer.
 - (b) Less than 5% of hospitalized older patients with end-stage cancer knew that the CPR success rates were <10%.</p>
 - (c) Physicians tend to rate more end-of-life items as significantly more important than patients.
 - (d) More than 60% of seriously ill older adults consider inability to "get out of bed" as a better quality of life than death.
- 185. Which of the following statements is *not* true regarding healthcare professional's concerns about family-witnessed resuscitation?
 - (a) Some hospitals may have inadequate resuscitation space per patient volume.
 - (b) Family-witnessed resuscitation causes additional stress to healthcare workers and thus negatively impacts their performance.
 - (c) Family members are not expected to understand the procedures and may interfere resuscitation efforts.
 - (d) Presence of family members may increase the potential for litigation in the future.
 - (e) The family members generally understand that resuscitation does not always result in positive outcome.
- 186. What is the most likely precipitating cause of in-hospital CPR in cancer patients?
 - (a) Anaphylactic reaction
 - (b) Respiratory failure
 - (c) Arrhythmia
 - (d) Cardiac arrest
 - (e) Hypovolemia
- 187. Which of the following conditions may have a worse outcome if a patient with such condition were to require CPR?
 - (a) Lymphoma

- (b) Breast cancer
- (c) Colorectal cancer
- (d) Genitourinary cancer
- (e) Head and neck cancer
- 188. Which of the following are considered by experts to be essential components of goals-of-care conversations in the ED?
 - (a) Identify the minimum quality of life the patient is willing to live for.
 - (b) Identify the surrogate decision maker if patient lacks capacity.
 - (c) Respond to emotion to reduce the anxiety/stress from discussing goals of care.
 - (d) Make a recommendation based on patient's baseline function, minimum quality of life acceptable, and prognostic estimates.
 - (e) Understand the baseline physical function and quality of life of the patient.
 - (f) All of above.
- 189. When should palliative care be started?
 - (a) Patients planning for tumor resections and starting chemotherapy
 - (b) Actively dying patients in ED
 - (c) Patients with chronic life-limiting illnesses such as congestive heart failure or chronic obstructive pulmonary disease
 - (d) When all cancer treatments have stopped
 - (e) A and C
- 190. Who would be the most effective ED palliative care champion?
 - (a) Medical students
 - (b) Quality control personnel
 - (c) Spiritual care personnel
 - (d) Hospital legal representatives
 - (e) ED volunteers
- 191. A 77-year-old man with advanced dementia, lung cancer with metastases to brain, presents from a nursing facility with increasing shortness of breath. He is bedbound and dependent on nursing care. There is a do not resuscitate/do not intubate order on his chart. His current vital signs are BP, 90/54; P, 105; R, 28; and T, 101.0 °F. Oxygen saturation is 89% on 4 L O₂ by nasal cannula. What is the most appropriate next step for patient management?
 - (a) Referral to in-patient hospice
 - (b) Intubation and admission to ICU
 - (c) Oxygen, antibiotics, and consult palliative care team
 - (d) Bereavement counseling
 - (e) Admit to hospitalist team for comfort care
- 192. In which scenario should an urgent consult to palliative care consult team be of most benefit in the ED?

- (a) A patient with severe malignant bowel obstruction who refuses surgical intervention
- (b) A patient with severe back pain and suspected acute cord compression
- (c) A new diagnosis of multiple metastases from a breast malignancy
- (d) A patient with severe vomiting after chemotherapy for lung cancer
- 193. In which scenario should hospice services referral be appropriately considered in the ED?
 - (a) A patient with dementia and prostate cancer
 - (b) Elderly patient with multiple myeloma and fall at home with hip fracture
 - (c) A patient with a chronic lymphocytic leukemia and a large left middle cerebral artery cerebrovascular accident or stroke
 - (d) Homeless patient with alcohol dependence and liver cirrhosis with a lung mass
 - (e) Any patient where expected survival is 6 months or less
- 194. A 43-year-old female has had evidence of worsening pain with known history of metastatic breast cancer to brain, lung, and bones. She has undergone radiation therapy as well as received steroids to manage her symptoms. You find that after repeated bolus IV doses of opioid analgesia, her pain remains uncontrolled and initiate a PCA. Her nurse tells you that she has improved; however, that she has been in the room frequently offering support and reassessing her symptoms. After clarifying goals of care, she has elected hospice with her family and her goal is to just get the pain under control. What level of hospice care does she likely require?
 - (a) Home hospice
 - (b) Facility-based care
 - (c) Continuous hospice care
 - (d) Inpatient unit care
- 195. A 75-year-old male has had progression of known prostate cancer and no longer has further traditional treatment options available per discussion with his oncologist after he presents to you with worsening back pain. After further discussion, he wants to pursue hospice, but remembers that there is a medication that has worked to provide longer life for patients in his situation and is off-label. He asks you if this medication will be covered by hospice?
 - (a) Yes, this medication is covered.
 - (b) No, it is not covered; it is a treatment that is meant to treat his underlying terminal condition.
 - (c) Call his oncologist to discuss.
 - (d) Look up the medication yourself to determine if it is effective or not.

- 196. A 50-year-old woman with a past medical history significant for recent diagnosis of metastatic pancreatic cancer not considered amenable to surgical intervention. She has experienced a rapid decline in clinical status, including obstructive jaundice, and due to this as well as complicating comorbidities of lupus with need for systemic immunosuppression, she was not considered a candidate for further therapies. She elected hospice and presents to the ED from home with bilious emesis. Further evaluation reveals likely SBO mechanical obstruction and she is not a candidate for surgical intervention. Her symptoms are temporized by a nasogastric tube (NGT), and she wishes to return home in lieu of admission but has severe ongoing nausea. What level of hospice is likely best able to control her symptoms at this point?
 - (a) She can return home; the hospice IDT should be able to manage her symptoms there.
 - (b) Continuous care should be able to transition her to home from the hospital.
 - (c) Respite care is the best option because her husband does not know how to manage a nasogastric tube.
 - (d) Inpatient unit care is most appropriate given her severe ongoing symptoms for stabilization prior to returning home.
- 197. Which of the following can emergency clinicians always rely on for appropriate guidance when faced with ethical dilemmas?
 - (a) Institutional and medical board policies
 - (b) The American College of Emergency Physicians' (ACEP) Code of Ethics
 - (c) The American Medical Association or American Osteopathic Association ethical statements
 - (d) None of the above
- 198. What are "values"?
 - (a) The standards that individuals, institutions, professions, and societies use to judge human behavior.
 - (b) The correct way of thinking about issues.
 - (c) A patient's normal view of medicine, society, and his or her family.
 - (d) A sense of duty to patients required of all healthcare workers.
- 199. Withholding treatment in the ED:
 - (a) Legally differs from withdrawing treatment.
 - (b) Morally differs from withdrawing treatment.
 - (c) Requires clinical information that is often unavailable immediately.
 - (d) Should never be done.
- 200. When using the "rapid approach to ethical problems" in the ED to decide on a course of action:
 - (a) Always consult with the bioethics committee/consultant before acting.

- (b) Assume that each ethical problem in the ED requires a unique solution.
- (c) Test your chosen action against your religious values.
- (d) When practicable and safe for the patient, buy time to consult on possible options.
- 201. Which one of the following statements is false?
 - (a) Members of the LGBTQ community can be more difficult to research due to lack of standard collection of sexual orientation and gender identity in the health record.
 - (b) Members of the LGBTQ community are less likely to live in poverty and more likely to have health insurance.
 - (c) Sexual orientation relates to identity rather than sexual behavior.
 - (d) Lack of employment discrimination protection is common, and in many states, you can be fired for being gay or transgender.
- 202. The minority stress model
 - (a) Only applies to racial and ethnic minorities
 - (b) Has not been shown to contribute to cancer disparities in minority groups
 - (c) Does not apply if all patients are treated equally
 - (d) Is thought to explain the negative effects of chronic stigmatization and high levels of stress on a minority group's health status
- 203. Cancers are with worse outcomes in the black community include:
 - (a) Colon cancer
 - (b) Breast cancer
 - (c) Lung cancer
 - (d) All of the above
- 204. Which of the following is not true?
 - (a) Minority populations will overtake the majority white population by 2050.
 - (b) Cultural humility is a lifelong process of selfreflection and self-critique that can inform understanding of cultural differences and how differences require sensitive approaches to healthcare.
 - (c) Outright discrimination is also known as implicit bias.
 - (d) Minority groups are often underrepresented in cancer clinical trials.
- 205. Which of the following is true regarding bias:
 - (a) Biases are shaped by experiences and based on learned associations between particular qualities and social categories.
 - (b) Raising awareness of physician bias and how they affect clinical decision-making is key to helping physicians mitigate bias.
 - (c) Patients with schizophrenia and other disabilities are often unable to advocate for themselves and

are also at increased risk of experiencing physician bias.

- (d) All of the above.
- 206. Choose the correct statement about WHO analgesic ladder:
 - (a) Provides a simple approach toward treating pain in only 50% of the patients.
 - (b) The first step of the ladder represents the use of nonopioid and opioids analgesics to treat mild pain.
 - (c) The second step represents the use of weak opioids with or without nonopioid analgesics, and with or without adjuvants to treat moderate pain.
 - (d) States to administer pain medications on-demand instead of around-the-clock.
 - (e) Intravenous and rectal dosing of drugs is preferred whenever possible.
- 207. Cancer pain can be characterized as:
 - (a) Neuropathic pain
 - (b) Somatic (nociceptive) pain
 - (c) Sympathetic pain
 - (d) A and C
 - (e) All of the above
- 208. Choose the correct statement about methadone:
 - (a) Rapid titration is safe.
 - (b) Withdrawal symptoms are more severe than morphine.
 - (c) Sedation and respiratory depression can outlast the analgesic action.
 - (d) Never used in opioid addiction.
 - (e) Lacks interactions with multiple medications that the patients may be on.
- 209. Which of the following is true regarding tramadol?
 - (a) Has a dose limit of 800 mg/d
 - (b) No effect on norepinephrine or serotonin
 - (c) Inhibits the reuptake of norepinephrine and serotonin
 - (d) Is a peripherally acting analgesic
 - (e) Has strong opioid characteristics
- 210. The specialties of emergency medicine and oncologic are organizing to establish professional groups to organize activities around oncologic emergency medicine, including educational, research, and policy initiatives. Which groups below are representative of these efforts?
 - (a) American Academy of Emergency Medicine Oncologic Emergency Medicine Interest Group
 - (b) Society for Academic Emergency Medicine Oncologic Emergencies Interest Group
 - (c) American College of Emergency Physicians Oncologic Emergency Medicine Section

- (d) NIH-supported Comprehensive Oncologic Emergencies Research Network (CONCERN)
- (e) B and D
- (f) A and C
- 211. Of the options below, which factors pose barriers to establishing quality metrics for oncologic emergency medicine?
 - (a) Gaps in existing ED measures
 - (b) Fragmented measure development
 - (c) Difficulty defining the episode of oncologic emergency care
 - (d) Measurement without a clear mechanism for improving ED care
 - (e) Challenges in obtaining ED quality data
 - (f) All of the above
- 212. The first patient navigator program was:
 - (a) Developed by the Dana Farber Cancer Center in 2005
 - (b) Developed by The University of Texas Cancer Center in 1994
 - (c) Developed by Harlem Hospital in 1990
 - (d) Developed by the Intermountain Healthcare system in 2002
- 213. In what year was Emergency Nursing established as a specialty practice by the American Nurses Association?(a) 1970
 - (b) 1986
 - (c) 1992
 - (d) 2011
- 214. In addition to the acute medical needs that prompt ED visits and invite the interventions of ED social workers, oncologic social workers serve to address the unique psychosocial stressors that accompany a cancer diagnosis. These may include:
 - (a) Adjustment to a new cancer diagnosis
 - (b) Alterations in role and identity
 - (c) Changes in caregiver needs and family roles
 - (d) Impact on work and finances
 - (e) Goals of care planning
 - (f) All of the above
- 215. All of the following statements regarding radiation are true, except:
 - (a) Radiation is the passage of an electromagnetic wave through space.
 - (b) Nonionizing radiation includes radio waves, microwaves, and ultrasound.
 - (c) X-rays and gamma rays are examples of ionizing radiation.
 - (d) Gamma rays are produced when electrons are emitted from electron clouds as a result of electron excitation.

- (e) Radiation is considered ionizing if it is of high enough energy to remove electrons from an atom.
- 216. Which of the following statements regarding lung cancer screening is true?
 - (a) Gays, lesbians, and bisexuals have higher rates of smoking, thus higher levels of lung cancer screening than their heterosexual counterparts.
 - (b) Geographically in the United States, the South has 40% of lung cancer screening-eligible patients and a screening rate that is one-third that of the Northeast.
 - (c) The NCI-funded National Lung Screening Trial (NLST) failed to demonstrate mortality reduction with lung cancer screening.
 - (d) The 1986 Mayo Lung Project found a benefit in overall mortality for subjects receiving lung cancer screening.
- 217. The most common tumor causing proptosis in adults is:
 - (a) Optic nerve glioma
 - (b) Lymphoma
 - (c) Orbital rhabdomyosarcoma
 - (d) Meningioma
- 218. Superior vena cava syndrome (SVCS) represents one of the most common oncologic emergencies seen in the ED.

- (a) True
- (b) False
- 219. All of the following regarding radioactivity are true, except:
 - (a) One Sievert (Sy) is equivalent to 1000 roentgen equivalent man (rem).
 - (b) One gray (Gy) is equivalent to 1 J of energy deposited in 1 kg of tissue.
 - (c) Radioactivity is measured in curies (Ci) in the English measurement system.
 - (d) Radioactivity is measured in Becquerel (Bq) in the SI system.
 - (e) A Becquerel (Bq) is equivalent to one disintegration of an atomic nucleus per second.
- 220. Which of the following are true statements?
 - (a) The antidote for fluoropyrimidine toxicity is oral capecitabine.
 - (b) Treatment of 5-FU toxicity is largely supportive.
 - (c) 5-Fluorouracil is the oral prodrug of capecitabine.
 - (d) Hand-and-foot syndrome is a characteristic adverse effect of uridine triacetate.
 - (e) Uridine triacetate toxicity is generally attributed to genetic polymorphisms.

Answers

- (b) Over 60% of patient with pancreas, brain, and lung cancer have at least one ED visit within 1 year of diagnosis. Overall, 44% of all cancer patients have one ED visit within 1 year of diagnosis. Less than 30% of those with melanoma, prostate, eye, or endocrine cancers have a visit within 1 year of diagnosis.
- 2. (a) Approximately 4% of all ED visits are cancerrelated, and lung, breast, prostate, and colon are the most frequent cancers associated with ED visits.
- 3. (d) Visit-level data only provide information about patients who visit the ED and do not capture all patients with cancer. While visit-level data may provide valuable information regarding the quality, type of care, and related diagnosis, visit-level data are unlikely account for multiple visits made by the same patient.
- 4. (e) Current methodologies generally utilize ICD codes and diagnoses to identify potentially avoidable ED visits; however, no common consensus definition exists to define which diagnoses or visits made by cancer patients are preventable.
- 5. (c) While specific visit and admission rates vary by cancer type, oncologic patients have higher rates of admission from the ED and higher rates of multiple ED visits than the general US population.
- 6. (e) HIPAA defined 18 potential identifiers. Of the above choices, gender is the least likely to uniquely identify a patient and does not meet these definitions.
- (c) A cancer registry is an information system that collects and analyzes data from a census of cancer cases. Registry data can be used to define and monitor cancer incidence, investigate treatment patterns, evaluate efforts to prevent cancer, and improve survival.
- 8. (e) Polling providers for preference, along with scouring the literature for recommendations and guidelines, are often first steps in designing an order set. Usually, one or more clinical "champions" are identified to begin the process of consulting literature, guidelines, experts in the domain, and practitioners in the affected departments.
- 9. (a) A current smoker is someone who has smoked at least 100 cigarettes lifetime and currently smokes every or some days.

- 10. (d) The patient is experiencing nicotine withdrawal after several hours of not smoking. She would benefit from nicotine replacement therapy. Combination treatments that include a short-acting medication (nicotine gum) supplemented by a long-acting medication (nicotine patch) are often more efficacious in treating nicotine withdrawal than patch alone.
- 11. (b) Both nicotine replacement therapy and motivational interviewing are efficacious. Tobacco control, in general, is among the most cost-effective treatments in all of medicine. National bodies do not mandate tobacco treatment in the ED, although the Model of the Clinical Practice of Emergency Medicine does require EM residents be taught the principles of tobacco dependence treatment.
- 12. (c) SBIRT does require smokers to be referred for post-ED treatment of tobacco dependence. Although tobacco pharmacotherapy may begin in the ED, SBIRT does not require it. Adding nicotine replacement to SBIRT is consistent with the newer treatment model known as Screening, Treatment Initiation, and Referral (STIR). SBIRT was initially developed to treat individuals with alcohol-use disorders, then adapted to tobacco and other substances. There are no contraindications to delivering SBIRT, although patients should of course be able to engage in a meaningful conversation with the interventionist.
- 13. (a) Alcohol use most commonly begins during adolescence. As youth transition into late adolescence, alcohol use typically increases.
- (c) Evidence supports a dose-dependent relationship between alcohol use and cancers of the breast, rectum, and larynx. Uterine cancer is not known to be closely linked to alcohol use.
- 15. (d) Native American populations have the highest rates of alcohol-attributable injuries such as MVCs and falls compared to other racial/ethnic minority groups. Black and Latino populations have the highest rates of recurrent or persistent alcohol dependence, once it has developed. Among all racial/ethnic minority groups, Asians populations have the lowest reported rates of alcohol use.

- 16. (b) The DSM-5 removed the category of alcoholrelated legal problems, which was present in the DSM-IV, and replaced it with the criteria of craving. Both tolerance and withdrawal were previous criteria for alcohol dependence in the DSM-IV. They are also criteria for alcohol-use disorder in the DSM-5.
- 17. (b) Total excision. All the other options may be useful for smaller lesions where total removal with the technique may be possible in depth and width, but risk base transection occurring or sampling error where the segment sampled of a lesion does not give an accurate reflection of its true nature (mild dysplasia abutting in situ melanoma in the same lesion). If uncertain regarding diagnosis or management, dermatology or teledermatology referral can be an alternative solution.
- 18. (c) The presence of a darker skin type on the Fitzpatrick scale is less likely to develop melanoma.
- 19. (b) Increasing Breslow thickness correlates strongly and negatively with 10-year survival.
- 20. (a) "A" in the ABCDE patient mnemonic stands for asymmetry, not aggravating.
- 21. (d) It is unclear if protection is life-long, but it is at least 10 years. Because the vaccines do not provide protection against all cancer-associated HPV types, routine cervical screening is still recommended, including routine screening in vaccinated women.
- 22. (c) Guidelines recommend screening for cervical cancer between the ages of 21 and 65. Cervical cancer screening should not be performed in women younger than 21 years of age, regardless of the age of onset of sexual activity. From ages 30–65, cervical cytology and HPV testing are recommended.
- 23. (c) Observation is preferred to treatment for women with CIN 1. In general, women with CIN2/3 are treated because of the higher risk of progression to invasive cancer. Treatment to remove abnormal areas of the cervix may be through ablation (with cryotherapy or thermal ablation) or excision of the precancerous area. Excisional procedures include loop electrosurgical excision procedure (LEEP), cold knife conization (CKC), and CO₂ laser conization. In the United States, excisional treatment is preferred to ablative treatment.
- 24. (d) Approximately 80% of individuals with HPV will clear the infection spontaneously within 18–24 months of infection. In women with persistent HPV infection, 3–5% will develop significant preinvasive disease, and <1% will develop cancer.
- 25. (d) This patient presents with a likely bowel obstruction from colon cancer. Important considerations include his history of ulcerative colitis and lack of subsequent follow-up. Long-standing ulcerative colitis is associated with a significant increased risk of colon cancer and regular screening for dysplasia is recom-

mended. Toxic megacolon may also present with systemic toxicity and colon dilation. Bloody diarrhea and distension are seen throughout the entire colon without a point of transition. Acute diverticulitis may have a systemic toxic presentation but distension is unlikely. While underlying ulcerative colitis is strongly associated with primary sclerosing cholangitis and ascending cholangitis, bowel obstruction is unlikely and a cholestatic pattern is seen with liver function tests.

- 26. (c) Occult fecal bleeding and hematochezia do not necessarily indicate the presence of advanced colon malignancy, and they are found in some individuals with early colorectal cancer.
- 27. (d) The patient has symptoms worrisome for colon carcinoma and will need a definitive diagnosis to guide therapy. CT colonography is a useful diagnostic method; however, the risk of radiation and need for further studies, such as colonoscopy, should be considered. Colonoscopy is the most accurate diagnostic method, can localize and provide a definitive tissue diagnosis, and can remove polyps.
- 28. (b) Upright abdominal series to look for free air. The patient has a history of colon cancer and suddenly developed abdominal pain suggestive of a bowel perforation. If she is unable to tolerate an abdominal series, a CT of the abdomen and pelvis would be the next step.
- 29. (d) Stabilization of a patient with hemoptysis and respiratory distress begins with establishing a protected airway in the form of endotracheal intubation. Once intubated, further steps to identify and isolate, such as with an endobronchial blocker or selective mainstem intubation, can be performed.
- 30. (c) Patients with greater dyspnea and poorer functional status derived the most benefit from therapeutic bronchoscopy based on data published from the AQuIRE registry. (Ost DE, Ernst A, Grosu HB, et al. Therapeutic bronchoscopy for malignant central airway obstruction: Success rates and impact on dyspnea and quality of life. *Chest*. 2015;147(5):1282–1298.)
- 31. (c) Patency of the lobar bronchi distal to the obstruction. Patency of distal airways, either on CT or on bronchoscopy, has been identified as predictors of a successful therapeutic bronchoscopy to relieve MCAO. (Ost DE, Ernst A, Grosu HB, et al. Therapeutic bronchoscopy for malignant central airway obstruction: Success rates and impact on dyspnea and quality of life. *Chest.* 2015;147(5):1282–1298; Giovacchini CX, Kessler ER, Merrick CM, et al. Clinical and radiographic predictors of successful therapeutic bronchoscopy for the relief of malignant central airway obstruction. *BMC Pulm Med.* 2019;19(1):219-019-0987-3.)
- 32. (b) Gas embolism is a reported complication of the use of argon plasma coagulation especially when used with

high flow rates and longer pulse durations. (Reddy C, Majid A, Michaud G, et al. Gas embolism following bronchoscopic argon plasma coagulation: A case series. *Chest*. 2008;134(5):1066–1069.)

- 33. (d) A hallmark of delirium is the fluctuation of attention levels, with repercussions on attention span, the ability to focus, and short-term memory. With multiple contributors to delirium, caregivers can miss this diagnosis.
- 34. (c) Patients with cancer on active treatment presenting with an acute ischemic stroke and meeting criteria for intravenous thrombolysis should receive this intervention in a shared decision process.
- 35. (c) Plateau waves are paroxysmal episodes, self-limited, with motor and autonomic manifestations that can be easily mistaken for seizures.
- 36. (a) It is rare to see spontaneous ICH with these coagulation parameters. She could have an incipient trauma, or hit her head on the floor after a fall (no clues in history) as the cause for the epidural hematoma. To preserve life and function, a hematoma evacuation is the definitive treatment in this case.
- 37. (b) Magnetic resonance imaging is the preferred imaging modality for MSCC diagnosis. The need for mobility assistance indicates a deficit that should be evaluated immediately. With a high suspicion for this patient, an MRI would be the appropriate next step to determine the definitive presence and extent of involvement of vertebral body metastasis. The need for mobility assistance indicates a deficit that should be evaluated immediately, rather than referred to outpatient pain management. Zoledronic acid is not used in treatment of MSCC.
- 38. (c) According to Bilsky's grading system, deformation of thecal sac without cord abutment is classified as Grade 1B. In the absence of mechanical instability, the initial treatment recommendation is radiation therapy. Radiation therapy alone has shown promising results for MSCC in maintaining functionality in the absence of neurological deficits. (Bilsky MH, Laufer I, Fourney DR, Groff M, Schmidt MH, Varga PP, Vrionis FD, Yamada Y, Gerszten PC, Kuklo TR. Reliability analysis of the epidural spinal cord compression scale. J Neurosurg Spine. 2010;13(3):324–8; Bilsky MH, Laufer I, Burch S. Shifting paradigms in the treatment of metastatic spine disease. Spine. 2009;34(22 Suppl):S101–7.)
- 39. (d) Loss of sensation, dense paraplegia, and incontinence are late findings of MSCC and signal some degree of permanent disability. While weakness and hyperreflexia are late findings of MSCC, urinary incontinence is the strongest indicator of poor prognosis and permanent disability.

- 40. (d) In spite of improved surveillance and diagnostic practices, 20% of malignant spinal cord compression occurs in patients without a known malignancy. A patient without a biopsy-confirmed cancer diagnosis in need of corticosteroid treatment presents a dilemma. If there is any question regarding the nature of the lesion, tissue diagnosis must be obtained without delay. Steroids are used with curative intent in treatment of plasmacytomas, thymomas, lymphomas, multiple myeloma, and germ-cell tumors. In these circumstances, corticosteroids given before tissue samples are obtained may cause regression of disease, hindering diagnosis and complicating delivery of definitive chemotherapy. In the absence of neurological deficit, corticosteroids may be withheld pending consultation with neurosurgery and oncologic.
- 41. (c) The most common hormonal deficit in pituitary apoplexy is a low level of growth hormone. The second most common is an acute deficiency of ACTH. Diabetes insipidus is possible, but uncommon, after pituitary apoplexy.
- 42. (d) Although the most common cranial neuropathy associated with pituitary apoplexy is dysfunction of the optic nerve caused by compression of tumor against the optic chiasm, this causes visual loss but not diplopia. Ocular motility is regulated by the III, IV, and VI cranial nerves, with the oculomotor nerve being the most vulnerable to pressure effects.
- 43. (a) Note that the hyperintensity on T1-weighted images lasts only until day 14, when it begins to darken (i.e., become hypointense) as hemoglobin is converted to methemoglobin. In T2-weighted images, where even more time-dependent fluctuation occurs, blood is hyperintense for 24 hours, then hypointense until day 7, then hyperintense for a week, and then after day 14, it has the same hypointensity seen on T1-weighted images after day 14.
- 44. (e) Use of a dopamine agonist (typically for a prolactinoma) is a predisposing factor to pituitary apoplexy in a small percent of patients, but not a dopamine antagonist. The other four factors listed are all even more common precipitators of apoplexy than is a dopamine agonist medication.
- 45. (c) This patient is likely presenting with advanced laryngeal cancer. These patients can decompensate due to airway compromise. The airway is best managed surgically in the operating room via awake tracheostomy, and attempted intubation in the ED should be avoided. In the meantime, intravenous steroids can be helpful. In an emergent situation, a "slash" tracheostomy or cricothyrotomy in the ED is warranted.
- 46. (b) This presentation is highly suggestive of a sentinel bleed from arterial bleeding into the pharynx. Patients

will typically present with brisk but short-lived bleeding from the oral cavity. The workup should be expeditious as bleeding can be catastrophic, particularly in the event of a carotid blowout. The next step in management would be CT angiography to assess for the site of hemorrhage and to guide potential endovascular intervention.

- 47. (d) A neck mass in an adult should be considered malignancy until proven otherwise. HPV-related oropharyngeal carcinoma in particular may present as a cystic neck mass with minimal symptoms from the primary site, typically in a male in their 50s–60s. A fine needle aspiration (FNA) can confirm the diagnosis. Incision and drainage should be avoided unless malignancy has been excluded.
- 48. (c) Facial paresthesia is not usually associated with acute sinusitis and suggests cranial nerve involvement, likely by an advanced sinonasal malignancy. Nasal congestion, drainage, and facial pressures are more typical of a patient presenting with acute sinusitis.
- 49. (c) Dasatinib has been linked to the development of pulmonary arterial hypertension, whereas nilotinib and ponatinib have been associated with an increased risk of myocardial ischemia, cerebrovascular accidents, and peripheral artery disease, in some cases requiring revascularization or leading to amputation. Of note, while the vascular toxicity of nilotinib may be related, at least in part, to its adverse effects on body weight and on glucose and lipid metabolism, the vascular toxicity of ponatinib appears to be a new form of microvascular angiopathy.
- 50. (b) Adverse cardiac events associated with ICI therapy are rare (less than 1% of patients) but can be potentially life-threatening. The most common and well characterized is ICI-induced acute fulminant myocarditis, which typically occurs within 90 days of initiation of therapy, and is often associated with acute hemodynamic failure and death in up to 50% of patients. The most commonly used and studied are nivolumab, pembrolizumab, and ipilimumab.
- 51. (d) Patients who develop systolic LV dysfunction (LVEF <50%) during anthracycline treatment should be treated with heart failure goal-directed medical therapy, including beta-blockers, angiotensin-converting enzyme inhibitors, or angiotensin II receptor blockers. The most studied agents are carvedilol, metoprolol, and bisoprolol, with notable recent trials showing significant reduction of mean change in LVEF using candesartan versus metoprolol and placebo. Statins, aspirin, enoxaparin, and rivaroxaban/apixaban have some clinical trial data to suggest cardioprotection as well. Dexrazoxane, a cardioprotective agent via iron chelation and topoisomerase II inhibition, may be used

if cumulative doxorubicin is $>300 \text{ mg/m}^2$ and/or the disease is metastatic.

- 52. (a) Antimetabolites such as 5-fluorouracil, clofarabine, and capecitabine; lomustine + bevacizumab + cyclophosphamide + etoposide; rapamycin, interferon alpha, and trastuzumab have demonstrated ability to cause cardiogenic shock secondary to stress cardiomyopathy, requiring ionotropic agents in up to 20% of patients. Acute stress cardiomyopathy usual occurs with the initial dose.
- 53. (d) Perform pleuroscopy and pleural biopsy and insert an indwelling pleural catheter (IPC). This patient has an undiagnosed lymphocytic exudative effusion that is recurrent in nature, which requires pleural biopsy to establish a diagnosis. The diagnostic yield of pleural biopsy is ~97%, compared to 62% after two thoracenteses. Given chest discomfort and presence of loculation, he has a nonexpandable, likely entrapped, lung. Talc pleurodesis in this setting will not be successful given loculation. Placement of IPC at the time of pleural biopsy is recommended to control the symptoms. Considering a repeat thoracentesis lacks utility and moreover could lead to additional loculations.
- 54. (b) There is no evidence that large chest tubes (20–24 F) are better than small bore chest tubes (10–14 F) in the management of pneumothoraces. Although the initial use of large bore chest tubes is not recommended, they may be necessary if there is a large air leak preventing complete re-expansion of the lung. The most common position of the chest tube is mid axillary line. In patients with persistent air leaks, one-way endobronchial valves are an alternative to surgical intervention.
- 55. (d) Hypercapnic acidosis reduces cyclic mechanical stretch, epithelial injury, and cell death. In ARDS patients with low tidal volume ventilation, permissive hypercapnia is accepted as increase in PaCo₂ and subsequent acidosis increases arterial and tissue oxygenation by a right shift of the oxygen-hemoglobin dissociation curve and possibly by increasing cardiac output and circulating catecholamines. Hypercapnic acidosis reduces cyclic mechanical stretch-induced nuclear factor-kB activation, reduces interleukin-8 production, and decreases epithelial injury and cell death compared to normocapnia. However, the rise of the PaCO₂ should occur gradually. Rapid rises should be avoided as the negative effects may exceed the beneficial ones (increased heart rate/blood pressure, arrhythmias, and pulmonary vasoconstriction/worsening hypoxemia).
- 56. (a) Nebulized tranexamic acid. In a randomized study by Wand et al., inhaled tranexamic acid (500 mg TID) showed a significant reduction in hemoptysis com-

pared to the control (96% vs. 50%) over 2 days. There were no adverse events related to the inhalation treatment. TXA when given as inhalation route is more efficacious because of the rapid onset of action compared to intravenous route.

- 57. (c) The patient is not low risk by Hestia criteria due to homelessness and poor social support. The patient would benefit from admission for social work.
- 58. (d) The patient demonstrates hemodynamic instability and would benefit immediately from fibrinolysis.
- 59. (d) The patient is likely Hestia low risk and can safely be discharged on a TSA (target specific anticoagulation).
- 60. (b) First-time VTE can be a presenting sign of occult, asymptomatic malignancy, but in a well-appearing patient with normal vital signs and no other complaints can be safely managed as an outpatient given Hestia low risk.
- 61. (g) The differential diagnosis is quite broad, and would include COPD exacerbation, COVID-19, bacterial pneumonia, immune pneumonitis, and pulmonary embolism. However, do not forget that immune checkpoint inhibitors can induce an insulinopenic diabetes mellitus that can lead to diabetic ketoacidosis. The patient would present with tachypnea and shortness of breath due to metabolic acidosis, along with nausea, dehydration, and general weakness.
- 62. (d) This patient has immune checkpoint inhibitorinduced hypophysitis. Headache is frequently associated with this immune-mediated adverse effect. This patient has central adrenal insufficiency and central hypothyroidism. Hypotension is due to adrenal insufficiency. Lack of reflex tachycardia is due to central hypothyroidism. Hyponatremia is due to both adrenal insufficiency and hypothyroidism. MRI of the sella will reveal pituitary enlargement. Hormonal measurements will reveal central adrenal insufficiency and central hypothyroidism. (El Majzoub I, Qdaisat A, Thein KZ, Win MA, Han MM, Jacobson K, et al. Adverse effects of immune checkpoint therapy in cancer patients visiting the emergency department of a comcancer center. Ann prehensive Emerg Med. 2019;73(1):79-87.)
- 63. (d) Numbness and tingling of extremities and muscle cramping are frequent symptoms of hypocalcemia. Complete thyroidectomy for thyroid cancer has a significant risk for postoperative hypoparathyroidism and the development of hypocalcemia. (Ramirez AT, Gibelli B, Tradati N, Giugliano G, Zurlo V, Grosso E, et al. Surgical management of thyroid cancer. *Expert Rev Anticancer Ther.* 2007;7(9):1203–14.)
- 64. (a) Stat EKG, and monitor cardiac rhythm while electrolyte abnormalities are corrected with intravenous

infusions since the patient has grade 4 hypomagnesemia and is at high risk for cardiac arrhythmia. Hypomagnesemia causes QT prolongation, and thereby increases the risk for torsade de pointes. Hypomagnesemia is caused by the nephrotoxic effect of oxaliplatin, but inhibition of EGFR in the renal tubules by panitumumab also causes hypomagnesemia. (Haroon N, Raza SM, Bhat ZY. Hypomagnesemia and chemotherapy, diagnostic dilemma, and treatment challenge: case report and literature review. *Am J Ther*. 2016;23(4):e1085–90; Oronsky B, Caroen S, Oronsky A, Dobalian VE, Oronsky N, Lybeck M, et al. Electrolyte disorders with platinum-based chemotherapy: mechanisms, manifestations and management. Cancer *Chemother Pharmacol*. 2017;80(5):895–907.)

- 65. (d) Often times, the bleeding tumor has several areas of active oozing, and mechanical disruption of the friable tumor with the use of a hemoclip may potentially worsen bleeding and is generally not effective for diffuse mucosal type bleeding. Hemoclips are generally avoided in bleeding from tumors unless there is a focal targeted area of hemorrhage, which is rarely the case.
- 66. (b) False. Overall prognosis should be considered prior to palliative luminal stenting. If patient life expectancy is anticipated to exceed 6 months, a more durable approach such as a surgical bypass (gastrojejunostomy) should be employed. While generally well tolerated, the main complications of gastroduodenal stenting include stent migration, perforation, and stent obstruction by tumor or food. These obstructions can typically be resolved endoscopically by placing a stent within the originally placed stent; however, the potential for re-intervention is higher after 6 months, and therefore, a surgical gastrojejunostomy should be considered in these patients.
- 67. (c) While TPN has been associated with elevated liver chemistries, it does not cause liver dysfunction, per se, and therefore is not a cause of liver failure. The rest of the listed causes are potential etiologies of fulminant hepatic failure.
- 68. (d) Patients with asymptomatic jaundice do not require biliary decompression unless their hyperbilirubinemia interferes with chemotherapy (i.e., some chemotherapeutic regimens require a normal bilirubin). Patients with intolerable jaundice or pruritus or poor nutritional status as a result of hyperbilirubinemia should have elective biliary decompression. Pruritus associated with hyperbilirubinemia can be debilitating and has been managed with antihistamines, corticosteroids, cholestyramine, and other medications with only limited success, and relief of obstruction is the mainstay of treatment. Those patients with signs and symptoms of acute cholangitis require urgent drainage and intravenous antibiotics.

- 69. (a) True. Nonoperative management consists of bowel rest and decompression using a nasogastric tube or long-intestinal tube, analgesia, and antiemetics, if necessary, and it has been showed to be effective in approximately 70–90% cases.
- 70. (a) Pancreatic cancer accounts for 15–20% cases, and it is the most common cause. It is often diagnosed when advanced, and it is the most frequent cause of gastric outlet obstruction.
- 71. (b) Neutropenic enterocolitis most commonly occurs2–3 weeks after receiving cytotoxic chemotherapy when neutropenia is most profound.
- 72. (d) Systemic sepsis is due to bacterial translocation through inflamed intestinal mucosa. Gastrointestinal bleeding is also common due to damage of the mucosa, which may lead to bowel obstruction.
- 73. (d) It is crucial to consider the diagnosis of immunemediated colitis caused by immune checkpoint inhibitors. The number stools per day puts the patient in Grade 3. A CT scan of the abdomen and pelvis is helpful in assessing bowel wall thickening and the extent of bowel involvement and exclude bowel perforation. Exclusion of infectious diarrhea is an integral part of the diagnostic process. The patient will need colonoscopy and biopsy. Empiric treatment with antimicrobials and glucocorticoid needs to be considered. (Rajha E, Chaftari P, Kamal M, Maamari J, Chaftari C, Yeung SJ. Gastrointestinal adverse events associated with immune checkpoint inhibitor therapy. *Gastroenterol Rep (Oxf)*. 2020;8(1):25–30.)
- 74. (c) These rare types of neuroendocrine cancer of the pancreas cause severe cholera-like diarrhea. The patients are often severely dehydrated with severe electrolyte abnormalities that are unlikely to be managed satisfactorily in the ED setting. Intravenous hydration and electrolyte replacement often require longer than 2 days. Additional intravenous octreotide would help to suppress tumor secretion of VIP. Chemoembolization to reduce tumor bulk often can reduce the symptom burden of diarrhea. (Dreanic J, Lepere C, El Hajjam M, Gouya H, Rougier P, Coriat R. Emergency therapy for liver metastases from advanced VIPoma: surgery or transarterial chemoembolization? *Ther Adv Med Oncol.* 2016;8(5):383–7.)
- 75. (d) Although the clinical scenario is consistent with acute graft-versus-host disease as the cause of severe diarrhea, infectious colitis and sepsis must also be excluded and empirical treatments need to be initiated immediately. (Robak K, Zambonelli J, Bilinski J, Basak GW. Diarrhea after allogeneic stem cell transplantation: beyond graft-versus-host disease. *Eur J Gastroenterol Hepatol.* 2017;29(5):495–502.)

- 76. (a) This patient has neutropenic fever and very probably also has neutropenic enteritis. CT scan of abdomen and pelvis is indicated. The epigastric pain can be from inflammation of the transverse colon. Hospital admission is indicated for neutropenic fever in patients with hematological malignancies. (Rodrigues FG, Dasilva G, Wexner SD. Neutropenic enterocolitis. *World J Gastroenterol: WJG*. 2017;23(1):42–7.)
- 77. (b) Metoclopramide is a prokinetic as well as an antiemetic and would be appropriate in this instance. The patient's nausea should be addressed before any oral laxatives are given. Ondansetron is known to cause constipation; therefore, it is not adequate in this instance. The patient has an unremarkable examination, and it is unlikely that she has a bowel obstruction. Obtaining a plain radiograph to confirm stool burden might be done, but the likelihood of important findings is low. Hydration with IV fluid alone is not appropriate as it is not likely to relieve her symptoms in the ED.
- 78. (a) Methylnaltrexone is FDA approved for OIC and could be used in the ED subcutaneously as it is fastacting. Senna and bisacodyl, as well as fiber, are less likely to work in the acute setting and may cause more abdominal discomfort without producing a bowel movement. Disimpacting the patient is not indicated if there is no evidence of impaction on rectal exam.
- 79. (c) Constipation can sometimes be the only symptom of cord compression. The subtle examination findings of lower extremity weakness with subjective paresthesias should alert the EP to possible spinal cord compression. An MRI of the spine should be ordered urgently. Giving pain medicine to the patients is adequate, but waiting for laboratory results before ordering the MRI is not appropriate. An enema may work, but constipation is not likely the primary cause in this case and missing cord compression can be catastrophic. A CT scan of the abdomen is not appropriate given the normal abdominal exam.
- 80. (c) This patient is ill appearing and has an examination consistent with peritonitis. He is also febrile and tachy-cardic. This patient requires immediate resuscitation and evaluation for free air by an emergent upright chest radiograph. While obtaining laboratory testing, ordering a lactic acid, and calling a surgical consult are all appropriate, they are not the most important next step.
- 81. (d) Drug-induced TMA manifests as a Coombs (-) hemolytic anemia, relative thrombocytopenia, unexplained renal failure, and de novo or worsening hypertension. Systemic manifestations such as low serum haptoglobin level, increased serum lactate dehydrogenase level, and schistocytes may also be present. All of the drugs listed are commonly associated with druginduced TMA, except ifosfamide.

- 82. (c) All of the listed interventions are useful in the treatment of MTX toxicity, except dialysis. While dialysis may clear MTX from the blood, there is a rebound effect as MTX if highly protein bound. Dialysis is not effective in removing drugs that are highly protein bound.
- 83. (a) Patients with SIADH have impaired water excretion due to inappropriate antidiuretic hormone release. Patients with SIADH will excrete all of the sodium content of normal saline, but will retain some of the water, thereby worsening hyponatremia. Patients suspected of possibly having SIADH should not be started on normal saline initially.
- 84. (a) Allopurinol will prevent the conversion of xanthine to uric acid, but does not help to metabolize already formed uric acid in the bloodstream. However, rasburicase will effectively metabolize uric acid into allantoin. The use of dialysis in the prevention of TLS is controversial. Bicarbonate base fluids may help prevent uric acid deposition in the kidneys, but may also increase the formation of calcium phosphate crystals and worsen kidney function.
- 85. (d) Sometimes patients will present with urinary retention or new urinary symptoms following prostatectomy catheter removal. Generally, patients should not have a urethral catheter inserted unless the issue is discussed with a urologist due to a concern for traumatizing or further disrupting a recent vesicourethral anastomosis.
- 86. (c) Clot irrigation can sometimes require 30–60 minutes, and a reasonable attempt should be made to clear the bladder of clot prior to initiating additional treatments. Once the bladder clot has been completely evacuated, consideration is given to initiate CBI. It is important to note that CBI should not be used unless manual evacuation has ensured that large clots have been evacuated from the bladder since these clots are unlikely to clear with CBI alone and may obstruct the catheter during passage.
- 87. (d) Upper urinary tract drainage is accomplished by ureteral stenting or nephrostomy tube placement. In the infected patient, randomized data have not supported any difference in meaningful outcome such as time to fever resolution. Ultimately, the decision on which to perform is very nuanced and dependent on patient factors. In the patient requiring urgent intervention, the best measure will depend on what is available first. It is known that there is significant regional variation in stent versus nephrostomy tube placement likely for this reason. Contraindications to each decompression method are first assessed. Nephrostomy tubes will generally be contraindicated in those with coagulopathy and may be more difficult in the patient who cannot

be placed prone, has morbid obesity, or lacks hydronephrosis. Ureteral stents will not be feasible in the patient who does not have easy endoscopic access to the ureter – in the oncologic setting, this is due to trigonal invasion of a large pelvic mass or due to prior lower urinary tract surgery.

- 88. (c) Patients presenting after partial nephrectomy with hematuria may sometimes have developed a pseudoaneurysm or arteriovenous fistula. These patients may present with hemorrhagic shock. In the stable patient, initial resuscitation should be commenced and a pseudoaneurysm/AVM can be ruled out with CT angiography. In the unstable patient, urologic consultation should be obtained immediately. Management should be coordinated with urology and will often involve selective angioembolization by interventional radiology.
- 89. (b) Patients with hypercalcemia often exhibit symptoms such as dehydration, uremia, muscle weakness, diminished reflexes, and cardiac arrhythmias.
- 90. (a) Febrile neutropenia (FN) is defined as a single temperature of ≥ 38.3 °C (101.0 °F) or a sustained temperature of ≥38.0 °C (100.4 °F) for at least 1 hour in a patient with an absolute neutrophil count ≤ 500 cells/mm³ or 1000 cells/mm³ and likely to fall below 500/mm³.
- 91. (c) Acute blood loss in a cervical cancer patient may require vaginal packing, volume replacement with crystalloids and/or blood products, and palliative radiation therapy. Invasive procedures such as percutaneous embolization and endoscopic procedures are reserved when noninvasive methods fail.
- 92. (c) Necrotizing enterocolitis can present with nausea, vomiting, abdominal distension, diarrhea, and hematochezia. On CT classically bowel thickening, mesenteric stranding, mucosal enhancement, and pneumatosis is seen. Supportive measures are the treatment of choice.
- 93. (d) The most common cause of bone lesions and subsequent pathologic fractures in patients >40 years old is metastatic disease. Imaging should include CT chest, abdomen, and pelvis to identify the site of primary disease. Fractures of the tibial diaphysis should be immobilized in a well-padded long leg splint that goes well above the knee in order to establish rotational control.
- 94. (a) Total hip replacements most commonly dislocate posteriorly. They result in a shortened, flexed, adducted, and internally rotated limb. Patients often describe standing from a seated position as the mechanism of dislocation. Pending medical stability, closed reduction is safe and effective in the ED. Periprosthetic fractures are not typically associated with standing from a

seated position, but rather, usually occur in the setting of trauma. Buck's traction has not definitively been shown to increase patient comfort.

- 95. (c) With a history of diabetes, cancer, and recent major surgery, the patient has three common risk factors for necrotizing fasciitis. These patients often present with altered mental status. With the laboratory values listed, she, at a minimum, has a LRINEC (Laboratory Risk Indicator for Necrotizing Fasciitis) score of 9 (≥6 is highly suspicious for the diagnosis). While necrotizing fasciitis often has soft-tissue gas evident on CT scan, this is not always the case. The sooner this patient undergoes surgery, the better her chances of survival.
- 96. (b) The patient's history and examination are consistent with compartment syndrome. Patients with leukemia are often thrombocytopenic and prone to significant bleeding in the absence of severe trauma. Pain with passive stretch, paresthesias, and firmness on palpation are all consistent with this diagnosis. Absent pulses are a late finding in compartment syndrome. A Stryker needle is not indicated in this case, as the patient is alert and has characteristic examination findings. Further, he has profound thrombocytopenia, and invasive testing could exacerbate the bleeding that likely caused this event.
- 97. (b) When a maculopapular drug rash presents with fever, lymphadenopathy, or facial edema, drug reaction with eosinophilia and systemic symptoms (DRESS)/ drug-induced hypersensitivity syndrome (DIHS) should be considered, and a workup for systemic involvement should be performed. The cutaneous eruption is typically maculopapular and rarely presents with purpura, vesicles, or pustules. The liver is the most common site of visceral involvement, but other systemic findings include interstitial nephritis, pneumonitis, myocarditis, arthritis, cytopenias, atypical lymphocytosis, thyroiditis, and cerebritis. The clinical manifestations typically begin 2-6 weeks after initial exposure to the causative medication necessitating a drug history of the past several months when DRESS is suspected.
- 98. (b) Over 20% of patients treated with anticytotoxic T-lymphocyte–associated protein 4 (CTLA-4) or antiprogrammed death 1 (PD-1)/programmed death ligand 1 (PDL-1), and up to 72% with anti-CTLA-4/anti-PD-1 combination therapy develop immune-related cutaneous adverse events (ircAEs), most commonly pruritus and MPR. These ircAEs may be associated with prolonged progression-free survival and overall survival. Unlike MPR to traditional medications, those to immunotherapy may develop at any time during or months after completion of therapy, with a median onset of 62 days (range 1–1676 days) after initiation

therapy. Cutaneous morphology of immune-related MPR may be indistinguishable from that due to traditional medications, or have overlap with lichenoid, psoriasiform, or eczematous rashes.

- 99. (d) Purpuric eruptions describe nonblanching skin lesions secondary to hemorrhage into the skin. Purpuric lesions may be flat (macular purpura), small and raised (palpable purpura), or larger and netlike (retiform purpura). It is important to recognize the features of macular purpura, palpable purpura, and retiform purpura as the differential diagnosis varies based on these morphologic differences. Macular purpura typically indicates hemorrhage into the skin secondary to low or dysfunctional platelets or vessel wall fragility in the absence of inflammation. Palpable purpura describes small, raised, nonblanching lesions most commonly found on the lower extremities. Palpable purpura is the classic skin manifestation for cutaneous small vessel vasculitis. Retiform purpura describes cutaneous lesions that have a netlike or stellate (starlike) pattern of purpura often with central necrosis or ulceration, reflecting damage to larger vessels with resultant cutaneous ischemia and hemorrhage. Damage to the vessel may occur either through infiltration of the vessel wall or occlusion of the vessel lumen.
- 100. (a) The SCORTEN severity of illness score for Stevens–Johnson syndrome and toxic epidermal necrolysis includes the following seven criteria: age >40 years, presence of malignancy, heart rate >20, percentage of epidermal detachment >10%, BUN >10 mmol/L, serum glucose >14 mmol/L, serum bicarbonate level <20 mmol/L. A score above 4 predicted >90% mortality.
- 101. (d) This patient presents with concern for typhlitis. The characteristic sign on ultrasound is a thickened bowl wall, of greater than 3 mm, which can be measured on ultrasound. The remaining imaging findings are all classic signs of other intra-abdominal pathology. The target sign is associated with intussusception, a non-compressible hyperemic blind-ended pouch is appendicitis, a periumbilical cystic structure may be a pancreatic pseudocyst, and the whirlpool sign is associated with ovarian torsion. Of these, typhlitis is the only treatment associated finding that would present with right lower quadrant pain.
- 102. (b) Patients with Burkitt's lymphoma, particularly those with high tumor burden at, are high risk for tumor lysis syndrome. This is characterized by elevations in potassium, phosphorus, and urea nitrogen. Calcium is typically decreased. There is no definitive effect on sodium.
- (b) While all of the above are potential treatment complications, most result in mortality during treatment,

whereas pediatric oncologic survivors have a higher than typical rate of suicide.

- 104. (b) Fatigue, bone pain, fevers, and pallor are all associated with pediatric cancer diagnoses. Weight loss, rather than weight gain, is also a particularly alarming findings in pediatric patients and requires further investigation.
- 105. (c) Initial management should begin with IV fluid. Although these patients may sometimes appear fluid overloaded with apparent pulmonary edema on chest X-ray, they are usually not hypervolemic, and IV fluids to mitigate the hyperviscous nature of the patient's blood is a key initial treatment. Antibiotics should not be the initial step in management; however, there should be a low threshold to start antibiotics, especially for gram-negative coverage. The patient will most likely be started on hydroxyurea, but initial treatment should be fluids. Patients with moderate-to-severe symptoms benefit from leukapheresis, which this patient may require. At this time, however, initial management is fluid resuscitation.
- 106. (a) This patient's presentation is concerning for leukostasis, but acute stroke and intracerebral hemorrhage are both high on the differential. A noncontrast CT head can help distinguish between etiologies and is rapidly available. A CT angiogram would not be appropriate as avoiding contrast is preferable in these patients with multiple insults to the kidneys, and whose renal function is likely to worsen with initiation of chemotherapy as tumor lysis syndrome ensues or worsens. In addition, the thrombocytopenia would contraindicate TPA. While brain MRI could also help establish a diagnosis, it takes much longer to obtain and is not as good as CT at identifying blood. This patient may require an MRI in the future if the CT scan is nondiagnostic, but this is not the best initial test. Meningoencephalitis is also on the differential diagnosis for this patient, especially as she may be immunocompromised. An LP could also help establish a diagnosis of subarachnoid hemorrhage. However, CT should precede LP in this patient for both diagnoses, as well as the more likely causes discussed above. Furthermore, this patient is severely thrombocytopenic, and ideally platelet counts should be a minimum of $20 \times 10^3 / \mu L$ prior to invasive procedures. If suspicion for meningoencephalitis is high, treatment should be started immediately and not delayed for LP. Seizure and/or postictal state is a less likely cause of altered mental status in this patient given her history, physical examination, and laboratory findings. EEG is of limited utility in the ED and would delay critical workup and management in this patient.
- 107. (d) This patient is experiencing hyperleukocytosis with myocardial leukostasis, and acute renal failure with

severe metabolic acidosis and compensatory tachypnea. In the setting of his renal failure, it is unlikely that his metabolic acidosis will markedly improve and he does not have much more room for additional respiratory compensation. Intubation and ventilation limit his body's metabolic demand and can help decrease the ongoing ischemia while awaiting leukapheresis. Matching the patient's preintubation minute ventilation on the ventilator is important to prevent loss of compensatory respiratory alkalosis causing worsening acidemia and potential hemodynamic decompensation. Although antiplatelets/anticoagulants are the appropriate management for non-ST elevation myocardial infarctions due to acute coronary syndrome, the likely etiology for this patient's ischemia is clogging of his coronary vessel(s) with leukocytes with additional metabolic demands from tachycardia and tachypnea. His thrombocytopenia is a relative contraindication to antiplatelet and anticoagulant therapies, and he is at a high risk for DIC. While consultation to cardiology would be appropriate, initiation of empiric antiplatelet/ anticoagulant therapies is inappropriate. Because the patient's main underlying cause of myocardial ischemia is most likely leukostasis, transfusion of packed red blood cells is not recommended as it will worsen the viscosity of the blood and further impair tissue perfusion. While the patient is likely functionally neutropenic and at risk for infection, his current presentation is not consistent with sepsis. Antibiotics may be appropriate if his chest X-ray demonstrates infiltrates or a focal infection is identified, but are unlikely to help his current presentation improve.

108. (b) Although without a clear focal deficit, this patient is altered and has a profound thrombocytopenia, as well as a high risk of DIC, that could lead to spontaneous intracranial hemorrhage that could potentially require neurosurgical intervention. STAT imaging of the brain is required. While this patient's tachypnea could very well be due to respiratory compensation for a metabolic acidosis given his low serum bicarbonate, his blood pressure is robust. While a sodium bicarbonate drip may be in his near future, it would be most appropriate to obtain a pH prior to initiating it in a nonemergent setting. A transfusion of platelets is appropriate in this patient; however, without knowing whether there is a CNS hemorrhage, it is unclear what his platelet goal should be. Transfusion of packed red blood cells, even if bleeding intracranially, is not appropriate as a contained intracranial bleed is not likely to be high volume, the patient is hemodynamically stable, and red blood cell transfusion will worsen blood viscosity and leukostasis. Although the patient's potassium is slightly high and his serum calcium is low, there is no indication that the patient has any signs of dangerous hyperkalemia, and his hypocalcemia likely exists in relation to his hyperphosphatemia.

- 109. (b) False: control of cancer-related factor VIII inhibitors does not required tumor control and immunosuppression should be started if tumor resection or other invasive procedures are initiated.
- 110. (c) L-Asparaginase is associated with low levels of multiple natural anticoagulants especially antithrombin and patients on this drug who have thrombosis should have their antithrombin levels supplemented.
- 111. (c) JAK2 in patients with myeloproliferative neoplasms is associated with increased risk of thrombosis. In addition, patients with visceral vein thrombosis often have JAK2 mutations – even with normal blood counts.
- 112. (a) Acquired deficiency of factor V is rare but can be seen in patients with myeloproliferative neoplasms a clue can be variable elevations of the INR/aPTT.
- 113. (d) The patient is presenting in aplastic crisis, with a low reticulocyte count and anemia as a result of a reduction in red blood cell production. An infection caused by parvovirus is often a source of aplastic crisis, and antibiotic therapy is not useful. Exchange transfusion is typically reserved for acute chest syndrome and does not increase the hemoglobin concentration. Oxygen is not beneficial unless the patient displays signs of hypoxia. The patient is most helped by a transfusion of red blood cells to improve her condition.
- 114. (a) Avascular necrosis is a complication of sickle cell disease. Progressive occlusion of microcirculation leads to increased intraosseous pressure and cell death. A fracture is more likely to occur in the setting of some trauma. Osteomyelitis typically presents with fever and may also have skin changes. The patient is otherwise young, and a tumor is not at an increased risk for tumor production.
- 115. (d) Laboratory workup is typically not required when the provider is familiar with caring for patients with sickle cell disease who are suffering from an uncomplicated pain crisis. However, laboratory testing should be considered when the patient is being admitted, another diagnosis is suspected, if the patient appears unwell, or if the provider suspects worsening anemia.
- 116. (a) The patient is presenting with acute chest syndrome, and the classic triad of acute chest syndrome includes fever, hypoxia, and pulmonary infiltrate. Acute chest syndrome often presents with chest pain, shortness of breath, and dyspnea on exertion. Occlusion due to sickling in the pulmonary vasculature causes the patient to become hypoxic, which drives more sickling to occur, which in turn causes more occlusion due to sickling.

- 117. (b) This patient presents with what appears to be acute lymphoblastic leukemia. It is not uncommon for a patient with acute leukemia to be seen in the ED prior to diagnosis. Although his overall white blood cell count may be elevated, his ANC is <500, and has been so for an unclear amount of time. This is severe neutropenia, and he is at risk for infection due to many pathogens. The patient's headache, ocular exam, and CT findings are all concerning for acute invasive fungal sinusitis. This is an infectious disease emergency and he needs rapid initiation of antifungal therapy. While an MRI may be obtained for characterizing and surgical planning, he needs urgent stabilization now. Both micafungin and amphotericin are antifungal therapies. Micafungin has activity against Candida spp. and Aspergillus spp. However, it notably does not cover other molds, such as the Mucorales. Amphotericin has broader antifungal coverage that includes molds and dimorphic fungi. Posaconazole, which was not an option, is also a reasonable empiric treatment for acute invasive fungal sinusitis. Finally, pathogenic diagnosis typically requires nasal endoscopy and sinus evaluation by a trained otolaryngologist. Nasopharyngeal swabs are unlikely to recover the offending pathogen and are not used for diagnosis.
- 118. (a) This is a neutropenic patient presenting with significant abdominal pain. The differential is broad but would include neutropenic enterocolitis, also known as typhlitis. Other considerations would be enteroinvasive pathogens such as E. coli or Shigella or other causes of colitis (viral, ischemic, etc.). Further, she notes recent decrease in diarrhea, which could be the natural course of illness but also raises flags for possible bowel obstruction, particularly in a patient with intra-abdominal malignancy. CT scan would help to narrow this differential and is the preferred modality for diagnosing neutropenic enterocolitis. In patients with neutropenic enterocolitis, antibiotics should be targeted toward gram-negative and anaerobic bacteria, which are commonly found in the gastrointestinal tract. gram-positive Resistant bacteria, including Enterococcus, typically are not involved. While some infections or illnesses will require G-CSF, this should not be a priority until the diagnosis and trend of neutropenia is known. Finally, while she is at risk for C. difficile colitis, this is a less common cause of hemorrhagic colitis.
- 119. (c) The patient is at high risk for central line-associated bloodstream infection (CLABSI) in addition to other infection syndromes associated with receiving cyto-toxic chemotherapy. Gram-positive bacteria including *S. aureus*, coagulase-negative staphylococci, and

Enterococcus are common pathogens associated with CLABSI. In addition, the patient is receiving total parenteral nutrition which places her at increased risk for CLABSI due to Candida species. A comprehensive workup for undifferentiated fever in an oncologic patient in the ED should begin with a complete blood cell count, metabolic panel, and at least 2 blood cultures from separate peripheral sites. Recognizing that peripheral access can be difficult to accomplish in some cases, one set can be drawn from the central venous catheter when necessary. Blood cultures obtained from catheters have traditionally been associated with higher contamination rates than those obtained from peripheral venipuncture. Obtaining 2 sets of blood cultures from the central venous catheter increases the risk of blood culture contamination, which can lead to an incorrect diagnosis of CLABSI and unnecessary removal of the central venous catheter, particularly if coagulase-negative staphylococci are isolated. It is appropriate to start empiric broadspectrum IV antibiotics, but after blood cultures have been obtained to guide diagnosis and treatment. As the patient is hemodynamically stable, additional workup for CLABSI can be completed prior to discussion with the patient's oncologist and deciding whether catheter removal is indicated to achieve source control.

- 120. (d) A history of high-dose corticosteroid use for GVHD coupled with the indolent onset of respiratory symptoms and lack of appropriate antibiotic prophylaxis raises the concern for *Pneumocystis jirovecii* pneumonia, which is supported by the chest radiograph findings. The patient reports compliance with taking valganciclovir, which provides effective prophylaxis against CMV and HSV infection. While he is not on prophylaxis for *Pseudomonas*, the typical presentation for Pseudomonas pneumonia would involve a more acute presentation and likely more localized infiltrates.
- 121. (d) There is no validated risk stratification tool for pediatric febrile neutropenia. Therefore, in the majority of cases, patients are admitted for a period of inpatient observation.
- 122. (c) It is suspected 80% of infections arise from endogenous flora.
- 123. (a) For safe discharge, a patient should live no more than 1 hour or more than 30 miles away from the clinic or hospital.
- 124. (c) Oral antibiotics are noninferior to intravenous antibiotics in cases of low-risk febrile neutropenia. This is true for both solid and hematologic malignancies.
- 125. (b) Evaluation for potential alternative causes should be performed in most cases. CT is recommended in this instance to help diagnose and characterize the

extent of disease as well as rule out other causes. Ultrasound cannot offer a definitive diagnosis.

- 126. (b) Persistent GI bleeding coupled with other clear surgical indications including abdominal free air. Involvement of the cecum is likely given its limited vascular supply; however, its involvement alone is not an indication for surgery. Degree of bowel wall thickening alone is also not an independent indication for surgery.
- 127. (c) Route of chemotherapy administration plays no role in neutropenic enterocolitis. All other elements are potential risk factors, especially timing from last chemotherapy (between days 12 and 17). Tumor types tend to be solid in adult and hematogenous in pediatrics.
- 128. (a) Recombinant granulocyte colony-stimulating factor (G-CSF) may be beneficial; however, data regarding its success have been mixed and it has not shown a reduction in mortality from neutropenic enterocolitis.
- 129. (b) Three domains of psychosocial model are the predisposing factors, precipitating factors, and perpetuating factors. Predisposing factor is the genetic makeup which makes the basis of vulnerability. Precipitating factor is the ongoing challenges, like substance misuse. Perpetuating factor is an acute event, such as loss of a loved one, that pulls the trigger.
- 130. (d) In addition to general risk factors of suicide, such as depression, they have the burden of adverse effects of cancer treatment. This may include physical symptoms such as intolerable pain and extreme fatigue. Cancer patients may face other challenges such as altered body image, role reversal, and loss of job which adds to their burden.
- 131. (a) Poor sleep, fatigue, and weight loss may be present in depression as well as in cancer patients undergoing treatment. Hopelessness and loss of interest in previously enjoyable activities are the key features of depression.
- 132. (c) First step in managing suicidal patients is his/her safety. Making sure he has no access to any weapon. Everything listed in the question is also essential. Patient should have a follow-up plan, needs an antidepressant and also a hot line number, but removing access to weapon remains the first step.
- 133. (c) Hypoactive delirium is more frequent than the hyperactive (agitated) type in cancer patients, and it is frequently missed. Most cases of delirium in cancer patients are of the mixed type, which has features of both, hyperactive and hypoactive. All other choices are correct.
- 134. (d) Although all are causes of delirium in cancer patients, medication's side effects are the most com-
996

mon. Opioids are the most common, followed by benzodiazepine and anticholinergic medications.

- 135. (d) Safety of the patient is the most important step in management of delirium as in the case discussed in the beginning of this chapter. Patients with delirium are at risk of hurting themselves or medical staff particularly nurses. Delirious patients may experience frequent falls and may pull IV lines.
- 136. (b) Refractory delirium is the most common reason for palliative sedation. The purpose of sedation is to control refractory symptoms and not to hasten death. The need to continue sedation should be evaluated frequently, and if symptom has improved, sedation should be weaned off or stopped. Palliative sedation should be differentiated from euthanasia.
- 137. (d) There are two types of transalar herniation: ascending and descending. Ascending transalar herniation occurs as a result of middle cranial fossa mass effect, leading to superior and anterior temporal lobe displacement across the sphenoid ridge. Descending transalar herniation results from frontal lobe mass effect, leading to posterior and inferior displacement across the sphenoid wing. Transalar herniation can lead to infarction in both ACA and MCA territories secondary to mass effect on the carotid terminus. Subfalcine herniation can cause ipsilateral ACA infarction, and uncal herniation can cause ipsilateral PCA infarction.
- 138. (e) Massive hemoptysis is defined as expectoration of ≥300–600 mL of blood within a 24-hour period. Pulmonary TB is the most common cause of massive hemoptysis worldwide. Bronchogenic carcinoma is the most common malignant cause of massive hemoptysis. Trauma, postoperative complications, and pulmonary fibrosis are not common causes of massive hemoptysis.
- 139. (d) Hounsfield unit is a relative quantitative measurement used to interpret radiodensity in CT relative to water, which has a HU of 0. Blood is slightly denser than water with a HU range of 30–60 (depending on its state, e.g., clotted, liquid, acute, subacute). Less dense materials such as air and fat have a HU of –1000 and –100, respectively. Bone is significantly denser, with cortical bone having an approximate HU of 1000.
- 140. (b) Radiographs are relatively insensitive to detecting lytic lesions compared to cross-sectional imaging such as CT or MRI. Radiographs can identify lytic lesions after loss of 30% or more of bone mineral density. Therefore, normal radiograph findings in setting of high clinical concern for a lytic lesion often warrant further evaluation with cross-sectional imaging.
- 141. (b) The patient described here presents with an examination and vital signs that are consistent with septic

shock. Visualized in the ultrasound image is a kidney which demonstrates hydronephrosis, given to be unilateral. In nononcologic patients, unilateral hydronephrosis is often associated with obstructive ureterolithiasis; however, in oncologic patients, it may also be associated with tumor compression of the ureter. The resulting backflow of urine to the kidney results in dilation of the renal calyx, which can be visualized directly with ultrasonography. Source control in this instance requires placement of either a ureteral stent or a nephrostomy tube to drain the obstructed kidney. Choosing dobutamine is incorrect, but would be considered if a bedside echocardiogram showed significantly decreased LV ejection fraction. A surgical consultation may be indicated in cases where the patient has free fluid within the abdomen suspected to be hemorrhagic in source, or if there were signs of malignant bowel obstruction. There is no indication of massive thromboembolism, which might be considered if bedside sonography shows septal flattening, McConnell's sign, or if the patient is known to have PE with decompensation. A CT should not be considered as the patient has unstable vital signs and would not benefit from awaiting an unnecessary imaging studies.

- 142. (e) The patient in this scenario presents with nonspecific symptoms common to the oncologic population. Historical features of his pain suggest pericarditis, but this description is not always consistent, and the classic findings of JVD, muffled heart sounds, and pulsus paradoxus are frequently missed on physical examination. His echocardiogram, however, clearly shows a pericardial effusion with right atrial inversion, a finding concerning for pericardial effusion with tamponade. Despite the patient's stable vital signs, he is at high risk for decompensation, and pericardiocentesis should be performed, ideally in a controlled environment with experienced operators, but in the ED if any instability develops. The other options describe the management of massive pulmonary embolus, another etiology of obstructive shock but with different management; ST-segment elevation MI, which is unlikely given the described ECG and patient presentation; pneumothorax and massive pleural effusion for which there is no evidence.
- 143. (b) The patient presents with unilateral leg swelling. There are multiple causes of edema but, given the asymmetric pattern seen here, a localized pathology rather than systemic pathology is more likely. Deep venous thrombosis (DVT) should be one of the first diagnosis to consider in an emergency setting especially in a patient with risk factors. The patient in this scenario has breast cancer and is undergoing chemotherapy which puts her at higher risk for venous throm-

boembolism. Additionally, she is presenting with leg swelling, tenderness, and overlying skin changes which makes DVT more likely. Using a risk stratification score such as the Well's score, the patient will be scored 4 out of 9, placing her as a high risk for DVT. One of the important sonographic characteristics for DVT is finding a noncompressible vein containing an intraluminal echogenicity which represents a thrombus. Of the choices given, B and C are the only describing venous findings. Of these two choices, B is consistent with DVT as the vein is found to be noncompressible indicating the presence of a thrombus.

- 144. (a) This patient has clinical signs of heart failure (hypoxia, JVD, symmetric bilateral lower extremity edema along with dyspnea) and is on a chemotherapeutic agent (doxorubicin) that is known to cause cardiotoxicity. The ultrasound findings of diffuse B lines, bilateral pleural effusions, and a decreased left ventricular ejection fraction coupled with her clinical signs and symptoms would confirm the diagnosis of heart failure.
- 145. (b) Duloxetine. Studies evaluating gabapentin and acupuncture have been conflicting or not shown improvement for chemotherapy-induced peripheral neuropathy (CIPN). Gel formulations have shown promise but the agent with the best evidence for use in CIPN is duloxetine. In 2013, a trial looking at CIPN caused by paclitaxel or oxaliplatin showed improvements in outcomes compared to placebo.
- 146. (a) Hyaluronidase is indicated for extravasation of vinca alkaloids. In addition, the arm should be elevated and a warm compress rather than a cold compress should be applied. Plastic surgery consultation is only indicated for a large volume vesicant extravasation; severe pain, if healing has not occurred 1–3 weeks after extravasation; or if there is early necrosis present.
- 147. (c) Epinephrine is recommended for the treatment of anaphylaxis in all current guidelines and injection into the thigh results in more rapid and higher plasma levels than other IM locations. Steroids and antihistamines are considered adjunctive therapy for anaphylaxis.
- 148. (b) SIADH with severe hyponatremia can occur after only one dose of cyclophosphamide. In addition, the patient is also on sertraline, which can also contribute to SIADH. She likely does not have brain metastasis and the recent MRI was negative. There are no signs of trauma or infection.
- 149. (d) The mechanism associated with the impact of radiation therapy on the heart and blood vessels is driven by inflammation related to the trauma of injury coupled with late fibrosis and functional instability. Intimal damage to cardiac vessels followed by cell proliferation in a disorganized manner can lead to premature

coronary artery disease and atherosclerosis. Damage to cardiac valves can lead to late fibrosis and calcification, which are able to cause both stenosis and insufficiency. Myocardial and pericardial inflammation can lead to muscle dysfunction, cardiomyopathy, and congestive heart failure. Pericarditis can be seen in situations where a large volume of heart receives radiation therapy and acute inflammation can lead to constriction of function. Electrical conduction deficits are seen and thought to be related to fibrosis in conduction pathways. Multiple chemotherapy agents influence cardiac health, and their relationship to injury imposed by

150. (d) Acute radiation effects to tissues are influenced by total radiation dose, daily treatment dose (fractionation), and volume of tissue treated. This information is often not immediately available to emergency departments when patients present for evaluation as specific treatment documentation is often in the department shadow record and not directly integrated into electronic healthcare records. The volume of the treatment target influences the number of stem cells directly affected from daily treatment. Daily treatment dose also influences injury to stem cells. Hence, total and daily doses as well as target volume are all directly related to the development of sequelae from the treatment.

radiation therapy remains under evaluation.

- 151. (a) True. There are reports of radiation injury to the lung tissue outside of the radiation treatment field. Although felt to be spurious at initial review, investigators have suggested that production of nitric oxide gas as a by-product of radiation-induced injury may play a role in generating injury in other parts of pulmonary parenchyma not directly in the radiation therapy treatment region.
- 152. (b) False. The mechanism of injury is multifactorial as radiation therapy likely has impact in all compartments of the kidney including the glomeruli, mesangium, and endothelium.
- 153. (a) True. At very high single-fraction total body doses (>10 Gy), death will occur through cerebrovascular syndrome in spite of support within 24–48 hours. The syndrome is due to uncontrollable swelling within the central nervous system associated with compromise of all neuromuscular processes. At total body doses of 5–12 Gy, death without support will occur in 1–2 weeks due to denudation and destruction of the gastrointestinal system associated with profound fluid loss and diarrhea. These cells have a self-renewal capacity measured within a few days; thus, a single total body dose of 10 Gy will eliminate a large portion of the stem cells within the gastrointestinal crypts. Although this dose does not affect differentiated adult cells, the exposure

eliminates the self-renewal potential of the stem cell; therefore, the mucosal surface of the gastrointestinal tract becomes denuded with no barrier for fluid and blood loss within a short period of time, measured in days. At total body exposure doses of 2-5 Gy, death will occur from damage to the hematopoietic system with primary damage to progenitor cells inhibiting self-renewal. Lymphocytes may die an intermitotic death, and this finding may be a surrogate biomarker for acute exposure within the first few hours to days of an incident. However, by day 30, most circulating blood elements are depleted with death often attributed to infection. The term, LD (lethal dose) 50/30, is borrowed from our pharmacology colleagues and reflects the LD of an agent that will cause 50% mortality in 30 days. Although radiation is not a drug, the LD 50/30 is now generally thought to be 5 Gy with modern hospital support.

- 154. (a) True. At total body exposure doses of 2–5 Gy, death will occur from damage to the hematopoietic system with primary damage to progenitor cells inhibiting self-renewal. Lymphocytes may die an intermitotic death, and this finding may be a surrogate biomarker for acute exposure within the first few hours to days of an incident.
- 155. (b) False. The use of bone marrow transplant in this setting remains controversial with strong advocates on both sides of the question.
- 156. (e) The pediatric population is at high risk of developing therapy-related mucositis. Ninety-nine percent of pediatric patients undergoing hematopoietic stem cell transplant will experience oral and/or gastrointestinal mucositis.
- 157. (c) The largest contributor to increased medical costs for patients with mucositis is extended inpatient hospitalizations, which can occur as a result of the potential sequelae of mucositis, such as infection or nutritional deficiency. A report based on data from 2012 estimated a combined cost of \$15,500 for every inpatient hospitalization due to severe mucositis.
- 158. (a) Benzydamine is currently the only agent for which there is level 1 evidence supporting its use in the management of mucositis. Benzydamine mouthwash can be used both for the prophylaxis and treatment of oral mucositis in patients receiving chemotherapy alone, radiation treatment alone, or concurrent chemoradiation to the head/neck.
- 159. (c) Significant risks associated with mucositis include severe infections, dehydration, and weight loss, requiring hospitalization. Such infections and severe malnutrition result in unexpected treatment breaks or premature discontinuation of antineoplastic therapies and are associated with increased rates of recurrence

and worse survival. Patients should be monitored closely with additional nutritional support including a feeding tube if needed and routine IV hydration.

- 160. (e) Leukoreduced and irradiated blood products should be given to all HCT recipients due to the risk of transfusion-associated GVHD, which is associated with high mortality.
- 161. (c) Splenic rupture is a rare but catastrophic complication of high-dose G-CSF.
- 162. (c) The patient's symptoms may be explained by mild acute GVHD (grade II) involving GI tract. Endoscopic studies should be performed to provide histologically corroboration the clinical suspicion and to rule out viral infection (particularly CMV).
- 163. (a) The most likely diagnosis is ICANS based on the clinical scenario. The ICE score is a 10-point score that can help the clinician to grade ICANS and further management based on grading.
- 164. (b) The differential diagnoses include acute coronary syndrome, pulmonary embolism, and immune-mediated myocarditis.
- 165. (d) Enasidenib is an inhibitor of isocitrate dehydrogenase 2 (IDH2) and can cause a differentiation syndrome that resembles sepsis. It is important to discuss with the oncologist and consider treatment with potent glucocorticoids.
- 166. (d) Blinatumomab is a bi-specific antibody that can cause an infusion reaction syndrome. Blinatumomab links T cells and malignant B cells and activates the T cells to exert cytotoxic effect on the malignant cells. Blinatumomab also causes a cytokine release syndrome. These leukemic patients are often neutropenic, and neutropenic fever is also likely.
- 167. (c) More than 1 ED visit in the last 30 days of life. Researchers used administrative healthcare data to describe this key indicator.
- 168. (d) 30%. It is impossible to have a system where the ED visit rate is zero, given it is entirely appropriate for individuals to visit the ED at the EOL for issues that arise; however, when administrative health data are used to examine systems and health policies, a rate of 30% is accepted as the benchmark for EOL ED visits.
- 169. (a) Fever. In a study that examined 18 EDs in the United States, pain, shortness of breath, nausea, and abdominal pain were found to be the most common ED diagnoses.
- 170. (c) Shorter survival. It is a common misconception that palliative care services lead to a faster death; however, studies show that access to early palliative improved quality of life and relieves suffering, without necessarily hastening death and even longer survival.
- 171. (a) Opioid analgesics have many potential side effects that may make patients or prescribers reluctant to use

the medications or use them in adequate doses to achieve pain control. Patients can develop a tolerance to certain side effects (e.g., nausea or respiratory depression); however, other side effects (e.g., constipation, pruritus) are not decreased with chronic use.

- 172. (b) Anticonvulsants such as gabapentin and pregabalin require close follow-up for titration. Peritoneal carcinomatosis generally causes visceral pain. Topical lidocaine patches should be worn for 12 hours and then removed for 12 hours. Corticosteroids are often used as adjuvants to treat a variety of pain types and have analgesic effects on pain from bony metastases and neuropathic pain.
- 173. (c) Opioids, with the exception of methadone, follow first-order kinetics and achieve their peak plasma concentration (and maximal analgesic effect) along a similar timeline: 60–90 minutes for oral/rectal administration, 30 minutes for subcutaneous/intramuscular administration, and 6–10 minutes for intravenous administration.
- 174. (c) For those patients in more severe pain, or requiring intravenous dosing for other reasons (difficulty swallowing), there are many available intravenous opioids. For the opioid-naïve patient, morphine is a safe, standard drug to start therapy. However, morphine should be used with caution in patients with renal impairment because one of the active metabolites (morphine-6-glucoronide) can accumulate with renal dysfunction. For intravenous dosing of the opioid-naïve patient, a starting dose of 2-5 mg of morphine (or equivalent) is recommended. This dose should be followed by a reassessment at 15 minutes, and if the pain score remains unchanged or increased, the initial dose given should be increased by 50-100%. If the pain score is decreased but still moderate (e.g., 4-7), the same initial dose should be repeated, and if the pain level is low (e.g., 0-3), then the initial dose can be used as needed.
- 175. (b) Pharmacologic agents to ease constipation are typically divided into five categories: bulk-forming agents, softeners, stimulants, osmotic agents, and peripheral mu-opioid receptor antagonists. Bulk-forming agents increase fecal mass to stimulate peristalsis. Stimulants act by increasing intestinal motility, whereas osmotic agents (e.g., polyethylene glycol, lactulose) act by increasing water content in the large bowel. The NCCN recommends prophylaxis with both a stimulant (or prokinetic) agent and the osmotic agent polyethylene glycol. Bulk-forming agents and stool softeners are unlikely to be effective in isolation. A 2010 Cochrane review recommended the use of polyethylene glycol over lactulose for chronic constipation because of better outcomes related to stool frequency, form, associated abdominal pain, and use of additional products. If

constipation persists despite the above medications, the provider can titrate the existing regimen or add an additional agent, such as magnesium hydroxide. Two peripherally acting mu-opioid receptor antagonists may be considered if laxative therapy has failed. Both methylnaltrexone, administered subcutaneously or orally, and naloxegol, an orally active agent, have demonstrated efficacy in reversing opioid-induced constipation.

- 176. (a) Younger age is a risk factor for substance abuse disorders.
- 177. (c) If a person engages in aberrant behaviors with their opioids in response to poorly controlled pain long enough, they can become actually addicted. In second-ary alcoholism, it is recognized that if a person uses alcohol to self-medicate panic disorder long enough, they can develop alcoholism in addition to their panic disorder.
- 178. (d) Only 30% had chart documentation suggesting a major need to improve recognition, diagnosis, and documentation of alcohol problems in people with advanced medical illness.
- 179. (b) Actually, longer drinking histories are predictive of a good response. When one sees these characteristics among others that predict a good response, it is clear why some patients undergoing cancer treatment who want to be able to adhere to their oncologic treatment may be good candidates for Antabuse.
- 180. (c) Morphine is the first-line agent for dyspnea in dying patients, particularly those whose goals are strictly focused on comfort. Intubating the patient is not in line with his expressed wishes and goals of care. BIPAP can be a useful tool at the EOL and can be considered, but morphine is still first line.
- 181. (b) Noninvasive ventilation can be a helpful tool for patients with dyspnea near the end of life, particularly if the underlying etiology of the patient's dyspnea is reversible. For some, it can help them through the acute crisis and improve dyspnea. In others, it is burdensome and may simply prolong dying.
- 182. (d) The etiology of dyspnea at the end of life is complex and often multifactorial. Some causes are treatable and reversible, while others are not.
- 183. (c) Palliative extubations are medically and ethically appropriate when the patient's condition and goals are not aligned with the use of mechanical ventilation. The process of extubation is relatively straightforward and within the purview of the emergency provider.
- 184. (b) Heyland et al. found that only 2.7% of hospitalized end-stage cancer or other advanced diseases understood that actual CPR success rates were <10%. (Heyland DK, Frank C, Groll D, Pichora D, Dodek P, Rocker G,

et al. Understanding cardiopulmonary resuscitation decision making. *Chest.* 2006;130(2):419–28.)

- 185. (e) The reality of what may occur during resuscitation differs than what is usually depicted in movies or on television in which many instances of resuscitations are shown to be successful.
- 186. (d) According to Giza et al., cardiac arrest is the most frequent precipitating cause in hospitalized patients requiring CPR. (Giza DE, Graham J, Donisan T, Balanescu DV, Crommet J, Botz G, et al. Impact of cardiopulmonary resuscitation on survival in cancer patients. Do not resuscitate before or after CPR? J Am Coll Cardiol CardioOncol. 2020;2(2):359–62.)
- 187. (a) Several studies show that in patients with solid tumors, unexpected cardiac arrest may be related to reversible problems, and those patients may be more responsive to CPR. (Giza DE, Graham J, Donisan T, Balanescu DV, Crommet J, Botz G, et al. Impact of cardiopulmonary resuscitation on survival in cancer Patients. Do not resuscitate before or after CPR? J Am Coll Cardiol CardioOncol. 2020;2(2):359–62.)
- 188. (f) All elements above are essential to goals-of-care conversations. (Ouchi K, Lawton AJ, Bowman J, et al. Managing code status conversations for seriously ill older adults in respiratory failure. *Ann Emerg Med.* 2020:S0196-0644(20)30410-8. doi: https://doi. org/10.1016/j.annemergmed.2020.05.039. Epub ahead of print.)
- 189. (e) Palliative care applies to *all* phases of a life-limiting condition and is not just for dying patients. In fact, maximal benefit is likely when there is early integration of palliative care into management plans as opposed to only considering such care as a last resort measure when "no more can be done" for the patient. The early integration of palliative care is associated with a higher quality of life, including better understanding and communication, access to home care, emotional and spiritual support, well-being and dignity, care at time of death, and lighter symptom burden. In fact, some evidence suggests that, on average, palliative care and hospice patients may live longer than similarly ill patients who do not receive such care. Palliative care also has the ability to simultaneously improve quality and control the cost of care for the most seriously ill patients.
- 190. (c) Spiritual care personnel are often fully initiated into palliative care services and serve as valuable champions to link existing resources to ED initiatives. It is important to begin an ED-PC initiative with identified ED "champions" who can effectively build upon lessons learned from other prior successes and failures so that the initiative is tailored to fit the unique ED setting. Integration initiatives in the ED may have a higher

chance of success when ED champions are fully engaged with palliative care experts to collaboratively define not only resources and processes but also appropriate metrics to track outcomes and measure impact of the integrated initiative.

- 191. (c) Oxygen and antibiotics are appropriate interventions while initiating efforts to define prognosis and goals of treatment.
- 192. (a) Bowel obstruction in patients with advanced cancer is rarely a surgical emergency and typically allows time to consider multidisciplinary aspects of the patient's condition. Palliative care specialists can provide valuable input into the appropriate goals of care. Previous treatment, cancer stage, and prognosis are unique variables to consider in cancer patients with gastrointestinal obstruction. Although accurate prognostication is difficult, it is helpful to attempt to determine if a patient can recover from abdominal surgery and obtain a meaningful quality of life prior to death from their malignancy.
- 193. (e) Patients may be referred for hospice services if their illness physiologically meets criteria of a prognosis of 6 months or less, should the illness run its usual course, and after counseling regarding the implications of this prognosis and available options, have accepted this prognosis and elect a treatment regimen and plan of care which focuses only on comfort-based treatments and services.
- 194. (d) The patient clearly has difficult to control symptoms requiring frequent nursing assessments indicating the need for inpatient unit level of care.
- 195. (b) However, a caveat should be made that there are some "bridge" programs that allow the patient to pursue treatment while receiving some hospice benefits. In addition, some commercial insurance plans may evaluate specific treatment requests in conjunction with hospice care. This is really where nuances can be quite intricate and worthwhile to explore directly with the medical director of the hospice of the patient's choice with palliative care team assistance.
- 196. (d) The patient has severe, ongoing symptoms that warrant further management with possible parenteral medications and ongoing frequent nursing assessments to stabilize her comfort and treatment regimen that would not be possible in a routine level of care. Continuous care is meant for the short-term stabilization of a new symptom which is likely to be controlled quickly, typically in 1–2 days; however, it is unlikely that she will have her symptoms stabilized in the setting of an acute obstruction with need for ongoing medication titration and assessment of NGT management. Calling the patient's hospice and reviewing her presentation and current goals would likely facilitate this transfer.

- 197. (d) All these examples are great for guidance on ethical dilemmas; none of them can "always" have guidance for every situation that can occur.
- 198. (a) Values are an acquired set of norms that individuals, groups, and organizations use as standards to measure others. While some values are pervasive, there is no uniform set of values across all individuals and groups.
- 199. (c) In an emergency, not all relevant information is available at the time important decisions must be made.
- 200. (d) Ethical problems can often take some time to work out, but a rapid approach rarely has enough time for a formal ethics consult. Buying some time so a reasonable solution can be derived is often the best that can be done.
- 201. (b) Sexual minorities consistently have less access to healthcare, are less likely to be insured, and are more likely to live in poverty.
- 202. (d) The minority stress model explains that minority groups experience chronic high levels of stress from stigmatization that can contribute to poor outcomes and disparities.
- 203. (d) For the three leading causes of cancer death in the United States, colon, breast, and lung cancer, survival rates for Blacks lag those for Whites.
- 204. (c) Provider and systemic bias can either be explicit or implicit. Explicit bias is often easier to recognize as the outward expression of prejudice. Implicit bias, the unconscious attribution of qualities or values to a member of a certain group, can be much more difficult to recognize.
- 205. (d) Many physicians have different exposures and backgrounds than the patients they serve. Unconscious attitudes and stereotypes affect behaviors and decisions in healthcare. In order to have greater awareness of bias in patient care, it is critical for physicians to receive education, feedback, and coaching on equitable practice. Patients with schizophrenia are 1.5–2 times more likely to die of their cancer than patients without mental illness. Patients with schizophrenia are also more likely to present at advanced stages of disease, are offered surgery at a lower rate, and receive fewer chemotherapy sessions. Patients with schizophrenia and other disabilities are often unable to advocate for themselves and are also at increased risk of experiencing physician bias.
- 206. (c) The second step of the WHO analgesic ladder includes opioid for mild-to-moderate pain as well as nonopioids and adjuvant drugs.
- 207. (e) A simple way to classify different types of pain is whether they are related to tissue damage (somatic nociceptive, visceral nociceptive) or nervous system disorders (neuropathic). The sympathetic nervous system may play a role in exacerbating pain through activation of the sympathetic nervous system.

- 208. (c) Methadone can be difficult to administer in the acute setting, given that its peak effect on respiration can occur later than its peak analgesic effect.
- 209. (c) Tramadol is a mixed-mechanism drug that acts as a weak mu-opioid receptor agonist, but also demonstrates some norepinephrine and serotonin reuptake inhibition.
- 210. (e) As of early 2020, neither the American College of Emergency Physicians nor the American Academy of Emergency Physicians have formed such groups.
- 211. (f) All are challenges to developing such quality metrics. Many of these factors stem from substantial shortcomings in the funding, oversight, and coordination of measure development and public reporting for cancer care.
- 212. (c) The first patient navigator program was created by Harold Freeman in 1990 at Harlem Hospital in New York. He focused on underserved women with breast cancer. The goals of the program were to expand access to cancer screening, improve clinical follow-up among medically underserved women through community outreach, and reduce the time between an abnormal test result and diagnosis and/or treatment. Eliminating barriers to health access, such as lack of insurance or cultural and communication barriers, were also critically important. The navigation program was remarkably successful.
- 213. (d) The American Nurses Association recognized emergency nursing as a specialty practice in 2011. As a challenging and unique profession, this clinical practice area prepares nurses to provide prompt interventions to stabilize or prevent further patient deterioration. The fast-paced, high-acuity setting commands refined critical thinking, clinical assessment, communication, and prioritization skills.
- 214. (f) Oncologic social workers routinely address all of these stressors for cancer patients.
- 215. (d) Gamma rays are emitted from unstable nuclei as part of radioactive decay.
- 216. (b) Geographically, the South has 40% of screeningeligible patients, but has the lowest density of ACRdesignated screening centers, and a 3.5% screening rate. In contrast, the Northeast has the lowest eligible percentage (15.5%) but the highest density and screening rate (10.1%). While gays, lesbians, and bisexuals have higher rates of smoking (thus higher proportions that are screening-eligible), they receive LCS at rates similar to their heterosexual counterparts. The landmark National Lung Screening Trial (NLST) demonstrated mortality reduction with lung cancer screening for the first time. The NCI-sponsored 1986 Mayo Lung Project found no benefit in overall mortality for lung cancer screening. Other contemporaneous NCI trials at

Memorial Sloan Kettering and Johns Hopkins also failed to find mortality benefits.

- 217. (b) The most common primary cancer affecting the orbit in adults is lymphoma. Orbital lymphoma can be the extranodal manifestation of systemic lymphoma or may be the only site of lymphomatous involvement. Other benign or malignant tumors that can cause proptosis include optic nerve glioma, meningioma, orbital hemangioma, sarcoma, and metastatic lesions. The most important cause of sudden and progressive proptosis in children is orbital rhabdomyosarcoma.
- 218. (b) False. Superior vena cava syndrome (SVCS) refers to the constellation of signs and symptoms that accompany the occlusion of the superior vena cava, either due to internal or external causes. Although it is commonly described as an oncologic emergency, it is not only nonemergent in the majority of cases but also increasingly found in patients who do not have cancer.
- 219. (a) Radioactivity is the term used to describe how much energy is being released by radioactive material in a given time. Radioactivity is measured in curies (Ci) in the English measurement system or the becquerel (Bq) in the SI system (SI = International System of Units or Système International). A Bq is equivalent to one disintegration of an atomic nucleus per second. A Ci is equivalent to 3.7×1010 disintegrations per second (dps) and is based on the decay rate of radium-226. The unit rad is often used in the English

system to describe the amount of ionizing radiation that is absorbed in a cell, tissue, organ, or the body. It is equivalent to 100 ergs of energy deposited in 1 g of tissue. The gray (Gy) is equivalent to 1 J of energy deposited in 1 kg of tissue. One Gy is equivalent to 100 rad. The rem (roentgen equivalent man) is a unit of equivalent dose which is used to measure the long-term biological risk related to ionizing radiation exposure (in the United States). The sievert (Sv) is the international unit (SI) for equivalent dose. One Sv is equivalent to 100, rather than 1000, rem.

220. (b) The fluoropyrimidine derivatives, 5-Fluorouracil (5-FU), and its oral prodrug capecitabine are mainstays of oncologic therapy. The therapeutic and toxicologic effects of 5-FU and capecitabine are attributed to their deleterious incorporation into RNA and DNA and to their inhibition of thymidylate synthetase. Hand-andfoot syndrome is a characteristic adverse effect of 5-FU and capecitabine. Toxicity may result from intentional overdose, iatrogenic overdose, or therapeutic administration. A significant proportion of severe toxicity occurs in the setting of therapeutic administration rather than acute overdose and is generally attributed to genetic polymorphisms. Treatment of toxicity is largely supportive. The antidote for fluoropyrimidine toxicity is uridine triacetate, but its indications are limited, and ideally, the treating oncologist or a toxicologist should be consulted prior to its administration.

Index

A

ABCDE mnemonic, 960 Abdominal distention, 93 Abdominal pain, 368, 369, 371 Abdominal symptoms, 90-93 nursing, 89, 93 Abdominopelvic emergencies, 600-605 biliary obstruction, 603 bowel obstruction, 600 abdominal radiographs, 600 intussusception, 602 ischemic colitis, 602 spontaneous pneumoperitoneum, 601 spontaneous intra-abdominal hemorrhage, 602 urinary obstruction, 602 Aberrant drug-taking behaviors differential diagnosis, 577 monitoring, 583 Abnormal mental status, 208 Abscess, orthopedics, 443 Absorption, 143 Angiotensin-converting-enzyme inhibitors, 275 Acetaldehyde, 131 Acetaminophen, 768 Acral lentiginous, 188 Acromegaly, 229 Acute abdomen adhesive disease, 344 clinical presentation and initial assessment, 353 diagnostic evaluation, 353 gastric outlet obstruction, 348 clinical presentation and initial assessment, 349 diagnostic evaluation, 349 treatment and operative intervention, 349 gastrointestinal bleeding, 349 clinical presentation and initial assessment, 350 diagnostic evaluation, 350, 351 treatment and operative intervention, 351 large bowel obstruction, 346 clinical presentation and initial assessment, 347 diagnostic evaluation, 347 treatment and operative intervention, 347, 348 malignant small bowel obstruction, 346 neutropenic enterocolitis, 351, 352 clinical presentation and initial assessment, 352 diagnostic workup, 352 treatment and operative intervention, 352 palliative treatment, 347, 348 radiation enteritis, 352, 353 radical cystectomy, 420 small bowel intussusception, 344 small bowel obstruction, 341, 342

clinical presentation and initial assessment, 342, 343 diagnostic workup, 343, 344 treatment and operative intervention, 344 treatment and operative intervention, 353 Acute angle-closure glaucoma, 264 Acute arterial bleeding, bleeding management, head and neck cancer, 251, 252 Acute blood loss, 426, 991 hemorrhagic cystitis treatment, 427, 428 invasive bleeding management, 427 management, 427 non-invasive bleeding management, 427 Acute chest syndrome (AChS), 534 anti-coagulation role, 535 blood transfusion, 535 hypoventilation, 535 initial management, 535 Acute coronary syndrome ACS CoC, 79 differential diagnoses, 998 Acute dacryocystitis, 264 Acute ischemic stroke assessment, treatment and prognosis, 217, 218 diagnosis, 217 importance and causes, 216, 217 Acute kidney injuries (AKI), 93, 397 and anti-VEGF therapy, 400 cancer patients, 397, 398 and conventional chemotherapy, 399, 400 definitions of, 398 hematopoietic stem cell transplantation and, 401, 402 and immunotherapy, 400 kidney disease improving global outcomes (KDIGO) criteria for, 399 multiple myeloma and, 401 Acute meningococcemia clinical manifestations, 465 diagnosis, 466 pathophysiology/etiology, 465 treatment, 466 Acute myelogenous leukemia, 970 Acute myeloid leukemia, 967, 978 Acute myocardial infarction (AMI), 46 Acute pancreatitis, 333 Acute pericarditis, 276 Acute promyelocytic leukemia (APL), 492-493 Acute radiation, 997 Acute radiation proctitis, 330, 426 Acute radiation syndrome (ARS), 545 concerted action, 545 cutaneous subsyndrome, 550 cutaneous syndrome, 546, 550

Acute radiation syndrome (ARS) (cont.) gastrointestinal subsyndrome, 550 gastrointestinal syndrome, 546, 551 hematopoietic syndrome, 545, 548 absolute neutrophil count, 548 ESA therapy, 548 **GM-CSF**, 548 HSC transplantation, 549 lymphocytes, 548 stem/progenitor cell, 548 TA-GVHD, 549 laboratory evaluation, 546 neurovascular/cerebrovascular syndrome, 546, 551 organ system involvement, 546 patient evaluation, 546 RITN, 549 goals, 549 IND detonation, 549 medical staff, 549 Acute respiratory distress syndrome (ARDS), 304 Acute respiratory failure (ARF), 303 Acute urinary retention, 413 Acute visual loss, 259-262 Adenocarcinoma, 163 Adenocarcinoma in situ (AIS), 165 Adhesiolysis, 430 Adhesive disease, 344 Adjuvant medications, 770, 771 Adrenal crisis causes, 390 diagnosis, 391 management, 391 symptoms, 390 Adrenal insufficiency, 387 Advance care planning (ACP), 97 Advanced practice providers (APPs), 18 Advanced trauma life support (ATLS) protocols, 441 Adverse cardiac events, 988 After-hours outpatient care, 45 Aggressive fluid resuscitation, 425 AI, see Artificial intelligence (AI) Airway obstruction, 248 Airway stenting, malignant central airway obstruction, 204, 205 AKI, see Acute kidney injuries (AKI) Alcohol, 131, 960 consumption, 129 intoxication, 134 regulation of, 130 types of cancers, 960 use, 129 Alcohol-attributable injury, racial/ethnic group, 960 Alcohol-related cancers, ED-based alcohol SBIRT, 135 Alcohol-related disease and injury, ED encounters for alcohol use, acute conditions related to, 130, 131 alcohol use, chronic conditions related to, 131, 132 health disparities, special considerations for, 132 Alcohol use disorder (AUD), 132 alcohol-related diseases and injuries, the identification and diagnosis of, 130 ED approach, 133, 134 identification and diagnosis of, 132, 133 Alcohol withdrawal, 580 delirium tremens, 580 guidelines, 581 medical treatment, 580-581 types and characteristics, 581

Wernicke-Korsakoff syndrome, 580 Allogeneic hematopoietic stem cell transplant, 361, 967 Allopurinol, 991 Altered mental status (AMS), 87, 88, 221, 555 central nervous system, 207 assessment and management, 208 causes, 208 diagnosis, 208 frequency and importance, 207, 208 diagnostic sequence, diagnosis of, 209 emergency department management of, 210 Ambulance diversion, 40 Ambulatory Care-Sensitive Condition (ACSC), 9 Amelanotic nodular melanoma, 188 American College of Emergency Physicians (ACEP), 77 American Medical Association's Physician Consortium for Performance Improvement (AMA-PCPI), 46 American Nurses Association, 1001 Amifostine, 312 Anaplastic thyroid carcinoma, 249 Anastrozole, 272 Andersen Behavioral Model of Health Services Use, 7 Angiogenesis inhibition by, 400 Angiosarcomas, 277 Angiotensin-converting enzyme inhibitors (ACEI), 400 Angiotensin receptor blockers (ARBs), 275, 400 Antabuse, 980 Anthracyclines, diastolic and restrictive cardiomyopathy, 279 Anticoagulation, 300, 318 Anti-EGFR targeted therapy, 383 Anti-metabolites, 988 Antivconvulsants, 999 Anxiety, diarrhea, 360 Apparent diffusion coefficient (ADC), 217 Arabinosylcytosine (ara-C), 263 ARDSnet PEEP, 305 Argon plasma coagulation (APC), 328, 330 malignant central airway obstruction, 204 ARS, see Acute radiation syndrome (ARS) Arterial thrombosis, 272 secondary to cancer, 272 secondary to cancer therapy, 272 Arthroplasty, orthopedics, 445 Artificial intelligence (AI), 190 Ascites, 93, 336 As low as reasonably achievable (ALARA), 146 Aspergillosis, 261 Asthma medication ratio, 56 Asthmatic-like reactions, 311 Asymmetric optic nerve disease, 259 Asymptomatic jaundice, 989 Atezolizumab-associated meningoencephalitis, 219 AUD, see Alcohol use disorder (AUD) Australia, burden of skin cancer, 186 Autoimmune thyroiditis, 391 Avascular necrosis, 972, 994 Awake tracheostomy, 251 Azygos vein, 292

В

Back pain, 89 Barrett's esophagus (BE), 329 Behavioral Risk Factor Surveillance System (BRFSS), 122 Benzydamine, 977 Bevacizumab, 310, 430 Bexarotene, 393 Bias, 982 Biliary and gastroduodenal stenting, 332 Biliary decompression, 334 Biliary metal stents, 334 Biliary obstruction, gastroenterology, 333-335 Billings algorithm, 9 Bilsky's grading system, 987 Binge drinking, 130 Bioethics, 865 advance directives, 871 beneficence, 874 refusing analgesia, 875 refusing lifesaving treatment, 875 withdrawing treatment, 875 withholding treatment, 874 bioethical dilemma, 872 committees and consultants, 874 confidentiality, 877 decision-making capacity, 870 capacity, 870 communitarianism, 870 components, 870 evaluation, 870 deontology, 869 distributive or comparative justice, 877 ethical theories, 867 future aspects, 879 impartiality test, 872 implied consent, 871 informed consent, 871 interpersonal justifiability test, 873 laws, 867 mid-level principles, 869 natural law, 869 nonmaleficence, 875 demands to "Do Everything", 875 research protocols, 876 oaths/codes, 867-869 presumed consent, 871 principles and virtues, 871, 874 professional and societal values, 865 rapid decision-making model, 872 religion, 867 surrogate decision-maker, 871 truth telling, 877 diagnosis notification, 877-878 survivor notification, 878 universalizability test, 873 utilitarianism, 869 values in emergency medicine, 866 virtue theory, 869 virtues in emergency medicine, 866 Biological effects, of absorbed radiation, 144 Bisphosphonates, 243, 384 Bitemporal visual field deficit, 229 Bivalent vaccine, 154 Bladder masses, 411, 412 Bleeding, 491-492 acquired von willebrand disease, 491 acute promyelocytic leukemia, 492-493 BTK inhibitor ibrutinib, 493 CML therapy, 493 dysproteinemia, 493 factor VIII deficiency, 491 immune thrombocytopenic purpura, 492

myeloproliferative neoplasms, 493 thrombotic microangiopathies, 492 thrombotic thrombocytopenic purpura, 492 Blinatumumab, 998 Blood alcohol concentration (BAC), 130 Bolus fluorouracil, 318 Bone marrow stem cell transplant, diarrhea, 359 Bowel function index, 370 Bowel obstruction, 425, 1000 colorectal cancer, 182, 183 constipation, 370 Bowel perforation, constipation, 370, 371 Bradyarrythmias, 283 Bradycardia, 283 Brain herniation assessment, 211 causes, 210 management and prognosis, 211, 212 principles, 209, 210 Breast cancer, 132, 135 Brief pain inventory, 766 Broad-spectrum intravenous antimicrobial therapy, 264 Bronchial artery embolization (BAE), 201, 202 Bronchiolitis obliterans with organizing pneumonia (BOOP), 312 Bronchoalveolar carcinoma (BAL), 165, 166 Bronchopleural fistulas (BPFs), 306 Bronchoscopy, 198, 200, 310, 312 Bronchospasm, 311 Budd-chiari syndrome, 417 Bulky inguinal adenopathy, 418 Bupropion, 124 Bursitis, orthopedics, 443

С

CAHPS® trademark, 41 Calcitonin, 404 Campylobacteriosis, 363 Cancer. 238 Cancer Emergency Department, 19-21 Cancer pain management, 983 challenges and opportunities, 893, 895 consumption in low- and middle-income countries, 898 etiologies, 892 neuropathic pain, 893 somatic nociceptive pain, 893 syndromes, 892 tumor burden, 893 visceral nociceptive pain, 893 interventional procedures, 899 non-opioid pharmacologic management, 896-897 non-pharmacologic management, 899 opioid pharmacologic management, 897-898 overview, 891 recommendations, 900-901 Cancer Registries Amendment Act, 34 Cancer registry, 34, 959 Cancer-related emergencies, 21 Cancer-related malnutrition, 387 Cancer Surveillance Programs, 34 Capecitabine, 1002 Cardiac arrest, 980 Cardiac computed tomography, 285 Cardiac dysfunction, 279 Cardiac magnetic resonance (CMR) imaging, 281, 285 Cardiac tamponade, 277

Cardiomyopathies diagnosis, 279 diastolic and restrictive cardiomyopathy, 279 immune checkpoint inhibitor-induced cardiomyopathy, 277, 279 stress cardiomyopathy, 277 systolic cardiomyopathy, 277 treatment, 280 Cardiopulmonary resuscitation (CPR), 789 comprehensive cancer center, 790 dying process, 789 ED cancer patients, 789 empathic and goal-concordant recommendation, 790 ethical considerations, 793 family members, 792 family-witnessed resuscitation, 791 healthcare workers, 792-793 palliative and supportive care, 790 patients, 793 Cardiovascular imaging, 284 cardiac computed tomography, 285, 286 cardiac magnetic resonance imaging, 285 echocardiogram, 285 Cardiovascular oncologic emergencies, 269 cancer therapies, cardiovascular toxicities of, 270 cardiomyopathies diagnosis, 279 diastolic and restrictive cardiomyopathy, 279 immune checkpoint inhibitor-induced cardiomyopathy, 277, 279 stress cardiomyopathy, 277 systolic cardiomyopathy, 277 treatment, 280 cardiovascular imaging, 284 cardiac computed tomography, 285, 286 cardiac magnetic resonance imaging, 285 echocardiogram, 285 chemotherapeutic agents, 274 conduction disturbances, 281-283 COVID-19 era, cardio-oncologic in, 286 ischemic cardiovascular manifestations, 269-272 arterial thrombosis, 272 coronary artery vasospasm, 273 peripheral artery disease, 273, 274 premature coronary artery disease, 272, 273 vasculitis, 273 myocarditis, 280 diagnosis, 281 presentation, 280, 281 treatment, 281 non-ischemic vascular conditions, 274 hypertension, 274, 275 venous thromboembolism, 275, 276 pericardial diseases, 276 pericardial effusion, 277 pericarditis, 276, 277 12-lead ECG, 271 valvular heart disease, 283 management of, 284 modern techniques, 284 pathophysiology, 283, 284 Care coordination, 41, 42 Caregiver advise, record, enable (CARE) Act, 42 Caregiver burden, quality measures, 41, 42 Caregiver distress, palliative social work, 112 Carotid blowout, bleeding management, head and neck cancer, 252

CAR-T cell neurotoxicity, 219, 221 Casefinding, 34 Catheter-associated thrombosis, venous thromboembolism, 322, 323 Ceiling effect, 768 Celiac plexus block, 360 Cellulitis clinical manifestations, 454 diagnosis, 454 orthopedics, 442 pathophysiology, 454 treatment, 455 Center for Medicare & Medicaid Innovation (CMMI), 45 Center for Medicare and Medicaid Services (CMS), 9, 41, 101, 147 Center to advance palliative care (CAPC), 821 Central airway malignant obstruction, 308 Central line-associated bloodstream infection (CLABSI), 994 Central nervous system (CNS), 207 acute ischemic stroke assessment, treatment and prognosis, 217, 218 diagnosis, 217 importance and causes, 216, 217 altered mental status, approach to patient with, 207 assessment and management, 208 causes, 208 diagnosis, 208 frequency and importance, 207, 208 atezolizumab-associated meningoencephalitis, 219 brain herniation assessment, 211 causes, 210 management and prognosis, 211, 212 principles, 209, 210 CAR-T cell neurotoxicity, 219, 221 cerebral venous sinus thrombosis diagnosis, 219 frequency, 218 presentation, 218 treatment and prognosis, 219 intracranial hemorrhage diagnosis, 216 importance and causes, 215, 216 presentation, 216 treatment and prognosis, 216 pembrolizumab-associated myopathy, 219 status epilepticus assessment, 214 causes, 213 definition and classification, 212, 213 management and prognosis, 215 Central retinal artery occlusion (CRAO), 262 Central retinal vein occlusion (CRVO), 262 Cerebral edema, 210, 335 Cerebral herniation, 213 Cerebral venous sinus thrombosis, central nervous system diagnosis, 219 frequency, 218 presentation, 218 treatment and prognosis, 219 Cerebrovascular accidents (CVA), 536 Certification of terminal illness (CTI), 822 Cervical cancer, 151 cervical intraepithelial neoplasia, 152 terminology of, 153 treatment of, 155, 156 colposcopy, 155

cytology, 154, 155 efficacy, 153 emergency department, cervical cancer prevention in, 156 HPV vaccination, 156 screening, 157 human papillomavirus, 152 challenges and barriers, 154 testing, 155 vaccination, 152 low and middle-income countries, prevention, 156 primary prevention, 151 purpose of, 151 recommended screening methods, 153, 155 risk factors, 152 safety, 153 bivalent vaccine, 154 nonavalent vaccine, 154 quadrivalent vaccine, 154 schedule and doses, 153 screening, 154 guidelines for, 155 Cervical dysplasia, 152 Cervical intraepithelial neoplasia (CIN), 152 CIN2/CIN3, 960 terminology of, 153 treatment of, 155, 156 Chemotherapy colorectal cancer, 183 diarrhea, 358 head and neck cancer, 254, 255 myxedema coma, 392 non-thrombotic pulmonary embolism, 313 superior vena cava syndrome, 298 Chemotherapy and radiation therapy (CRT), 254 Chemotherapy-induced constipation, 369 Chemotherapy induced enterocolitis, 969 Chemotherapy-induced nausea and/or vomiting (CINV), 645 clinical features, 645 emetogenic agents, 645 pathophysiology, 645 pharmacologic treatment, 647-648 treatment, 646 Chemotherapy-induced peripheral neuropathy (CIPN), 639, 640 acupuncture, 641 characterstics, 641 epidemiology, 640 gabapentin, 641 pathophysiology, 640 pregabalin, 641 presentation/diagnosis, 640-641 prevention, 641 treatment, 641 Chemotherapy-related pulmonary toxicity immune check point inhibitor related pulmonary toxicity, 310 lung toxicity, 311 Chemotherapy toxicity, 637 anaphylaxis, 651-653 adjunctive therapy, 653 chemotherapeutic and targeted agents, 652 clinical Features, 652 epinephrine treatment, 652 glucocorticoids, 653 pathophysiology, 651-652 treatment, 652-653 CINV, 645

clinical features, 645

emetogenic agents, 645 pathophysiology, 645 pharmacologic treatment, 647-648 treatment, 646 CIPN, 639 acupuncture, 641 characterstics, 641 epidemiology, 640 gabapentin, 641 pathophysiology, 640 pregabalin, 641 presentation/diagnosis, 640-641 prevention, 641 treatment, 641 CNS toxicity, 637, 638 busulfan, 639 cytarabine, 638 diagnosis and treatment, 639 ifosfamide, 639 methotrexate, 638 platinum-based compounds, 639 electrolyte disorders, 649, 650 cetuximab, 651 Cisplatin-induced hypomagnesemia, 649 cyclophosphamide, 649 ifosfamide therapy, 651 pathophysiology, 649 platinum-based, 649 treatment, 651 vinca alkaloids, 651 extravasation, 653 agents, 653 clinical Features, 653 dexrazoxane, 656 DMSO, 656 DNA binding vesicants, 655 hyaluronidase, 657 Non-binding DNA vesicants, 655 non-pharmacological and pharmacological treatments, 655 risk factors, 654 sodium thiosulfate, 657 surgical intervention, 657 treatment, 656 hypersensitive, types, 652 hypertension, 642, 644 acute aortic dissection, 644 acute coronary syndrome, 645 aggressive blood pressure control, 642 antihypertensive agents, 643 chemotherapy and targeted agents, 642 clinical features, 642 comorbidities and blood pressure, 643 heart failure, 644 hemorrhagic stroke, 645 ischemic stroke, 644 pathophysiology, 642 renal failure, 645 treatment, 642 nephrotoxicity, 646 alkylating agents, 648 antimetabolites, 648-649 antitumor antibiotic, 648 platinum-based compounds, 648 risk factors, 648 targeted therapy, 649 Chernobyl nuclear accident, 142

Chest emergencies central airway obstruction, 595 esophagorespiratory fistula, 595 massive hemoptysis, 598 pericardial effusion, 599 pericardial tamponade, 599 pulmonary embolism, 598 SVCS, 597 Chest pain, 84-87 Chest X-ray, lung cancer, 164, 165 Cholangitis, 333 Chronic back pain, 972 Chronic kidney disease (CKD) in cancer patients, 400, 401 Chronic obstructive pulmonary disease (COPD), 162, 377 Chronic osteomyelitis, 444 Chronic radiation proctitis, 350, 426 Chronic urinary retention, 413 Chyle leak, 253 CIN, see Cervical intraepithelial neoplasia (CIN) Cirrhosis, 131 Cisplatin-associated VTE, 275 Classical pituitary apoplexy, 226 Clinical decision rules (CDRs), 146 Clinical decision support (CDS), 25-29 ionizing radiation appropriateness criteria, 146 clinical decision rules, 146 computerized interventions, 146, 147 quality metrics and regulatory efforts, 147 shared decision-making with patients, 147 Clinical informatics, 23 cancer registry, 34 cancer surveillance programs, 34 clinical decision support, 25-28 communication order sets, 32, 33 transition of care tools, 33, 34 information technology systems Health Insurance Portability and Accountability Act, 32 telehealth, 30, 31 quality improvement, 24 workflow process redesign and quality improvement, 23, 24 analysis, 24 redesign, 24 Clinical manifestations, 452 Closed loop obstruction, 343 Clostridium difficile, 359, 361, 363 Clot, 319, 323 irrigation, 991 CMS State Operations Manual, 102 Coagulopathies, 322, 351 Co-analgesics, 770 Cognitive behavioral therapy, palliative social work, 113 Colitis, 351, 358 Colon cancer, 961 Colonic stents, 332 Colorectal cancer (CRC), 179 bowel obstruction, 182, 183 chemotherapy and radiation, 179 complications, 183, 184 correlates, stages of, 180 diagnosis, 180-182 recommendations, 181 TNM stage, 180 incidence and mortality reveal overall declining rates, 179, 180 lifetime risk, 179

oncologic emergencies associated with, 182 rectal bleeding, 183 perforation, 183 rectal bleeding, 183 symptoms of, 961 TNM categories and stage, 180 Colposcopy, cervical cancer, 155 Communication, 831 actively dying patient, 835-836 building blocks, 833-834 capacity, 833 conversation templates, 834 physical space, 833-834 preparation, 833 responding, 834 with children, 840 cultivate patient-centered care, 841 cultural considerations, 835 death of child, 839-841 disagreements among physicians, 838-839 dying in isolation, 839 emergency physician, 831 functional status, 831 language, 835 manage requests, 838 order sets, 32, 33 palliative physicians, 831 physicians, 831 prognosis, 831 progression of disease, 836-837 reflection and engagement, 841 REMAP guide, 834 at sign out, 840 silence, 834 spiritual and religious considerations, 835 sudden/unexpected death, 837 telephone death notification, 839 transition of care tools, 33, 34 Community Oncologic Medical HOME (COME HOME), 45 Compartment syndrome, 970 orthopedics, 441, 442 Complex case management, palliative social work, 110 Comprehensive end-of-life care initiative, 44 Comprehensive metabolic panel (CMP), 87 Comprehensive oncologic emergencies research network (CONCERN), 928-929 Computed tomography (CT) of brain, 230 cumulative radiation, contribution to, 139, 140 increased utilization, reasons for, 140, 141 lung cancer, 165 repeat/multiple CT imaging, patterns of, 141 superior vena cava syndrome, 295 transformative tool, medicine, 139 utilization, explosion, 139 Computerized provider order entry (CPOE), 31 Condition for Coverage (CfC), 101 Conditions of Participation (CoPs), 101 Conduction disturbances, 281-283 Cone-tipped (Toomey) syringe, 409 Confusion, 555 Conjunctivitis, 263 Consolidated Omnibus Budget Reconciliation Act (COBRA), 38, 101 Constipation, 367, 990 abdominal radiograph, 367 clinical presentation, 369, 370

diagnosis, 372 differential, 370 bowel obstruction, 370 bowel perforation, 370, 371 diverticulitis, 371 hypercalcemia of malignancy, 371, 372 hypothyroidism, 371 malignant spinal cord compression, 371 disposition, 373 emergency department visits, 368 examination, 370 healthcare costs, constipation-related symptoms, 368 medications, 369 pathophysiology, 368, 369 peripheral mu-opioid receptor antagonists, 770 prevention and patient education, 374 Rome Diagnostic Criteria, 368 treatment, pharmacological agents, 372 enemas, 373 lubiprostone, 373 methylnaltrexone, 373 naldemedine, 373 naloxegol, 373 osmotic laxatives, 373 secretagogues and peripheral acting µ-opioid receptor antagonists, 373 stimulant laxatives, 373 stool softeners and lubricants, 373 Consumer Assessment of Healthcare Providers and Systems (CAHPS), 41 Continuous bladder irrigation (CBI), 410, 969 Contrast induced nephropathy (CIN), 319 Contrast venography, 295 Conventional chemotherapy related pulmonary toxicity, 311, 312 Convulsive (CSE), 212 Convulsive status epilepticus, management guidelines for, 215 Coordination of care, 79 Coronary angiography, 270 Coronary artery vasospasm, 273 Corticosteroid therapy, 312 Corticosteroids, 241 Cost-effectiveness analysis, 26 COVID-19, 99, 935 cases, 931 consensus measures, 940 economic considerations, 942-943 ED palliative care, 946, 947 epidemiology, 935-936 health disparities, 936 prevention, 100, 941-942 respiratory distress algorithm, 938 SARS-CoV-2, 935 screening and safety measures, 100 system impacts, 939-941 treatment, 937-939 workforce mental health, 944-946 Cranial nerve III dysfunction, 228 Cranial nerve palsies, 262 Cranial neuropathy, 963 Craving, 960 Creatinine, 407 Crisis intervention, palliative social work, 110, 113, 114 Cryoprobe, malignant central airway obstruction, 204 Cryospray, malignant central airway obstruction, 204 Cryotherapy, 329 Crystalloids, 351

CT angiography (CTA), 295 CT dose index (CTDI), 144 Current smoker, 985 Curriculum development, 903 educational topics, 905 emergency medicine residency educational strategies, 905-907 evaluation, 909, 910 feedback, 909, 910 general needs assessment, 904 goals and objectives, 905 implementation success, 908 problem identification, 904 fellowship training, 912 medical school education, 910, 911, 913 medicine fellowship, 913 medicine textbooks, 904 oncologic emergency, 903 Cushing syndrome, 229, 389 causes, 389 diagnosis, 390 management, 390 symptoms, 389 Cyclin-dependent kinase (CDK) 4/6 inhibitors, 359 Cyclooxygenase (COX), 768 Cyclophosphamide, 977 Cystoscopy, 407 Cytogenetic biodosimetry, 547 Cytokine release syndrome (CRS), 96

D

Dacryocystorhinostomy, 264 Dasatinib, 273, 988 D-dimer. 319 Deep venous thrombosis (DVT), 626, 996 color and spectral Doppler, 627 common femoral vein, 627 compression ultrasonography, 626 greater saphenous vein, 627 Deficit Reduction Act of 2005 (DRA), 46 Dehydration, 362, 420 Delayed hospice referral, 44, 45 Delirium, 133, 207-209, 217, 555, 975 agitation and hallucinations, 559 assessing the patient, 558 causes of, 975 central neurotransmitter disturbances, 556 chlorpromazine, 559 conditions, 556 definition, 555 Diagnostic and Statistical Manual of Mental Disorders, 557 diagnostic tools, 558 differential diagnosis, 558 disinhibition, 560 environment, 558-559 epidemiology, 555-556 hallmark of, 987 haloperidol, 559 management of cancer patient with, 975 medications, 556, 559 neurotoxicity, 556 palliative sedation, 560 patient evaluation, 557 patient psychomotor activity and arousal level, 557 prognosis, 561

Delirium (cont.) screening tools, 557-558 underlying cause, 560 Brain Tumor or Metastasis, 560 dehydration, 560 hypercalcemia, 560 hypoxia, 560 interventions, 560 opioid toxicity, 560 sepsis, 560 Depression, 8, 974 Dermoscopy, 187 Desmopressin (DDAVP), 234 Det Norske Veritas (DNV), 101 Deterministic effects, 141 Diabetes insipidus, 378 Diabetes mellitus type 2 (DM2), 386 Diabetic ketoacidosis (DKA), 96, 387 Diarrhea, 357 causes, 357, 358 celiac plexus block, 360 diagnosis, 360-362 enteral feeding, 360 infectious enteritis, 359, 360 laxatives/magnesium-containing antacids, 360 management, 362 complicated chemotherapy-induced diarrhea, 363 dehydration and electrolyte imbalances, correction of, 362 GVHD, 364 immune-mediated colitis, 364 medication adjustment, 362 neutropenic enterocolitis, 364 paraneoplastic diarrhea, 364 pharmacologic therapy, 362, 363 targeted therapy, 364 treatment-induced diarrhea, 363 National Cancer Institute grading of, 360 paraneoplastic syndromes, 357, 358 stress and anxiety, 360 symptoms, 360 treatment-induced diarrhea bone marrow stem cell transplant, 359 chemotherapy, 358 immune checkpoint inhibitor therapy, 359 radiotherapy, 359 surgery, 359 targeted therapy, 358, 359 Diastolic cardiomyopathy anthracyclines and monoclonal antibodies, 279 radiation therapy, 279 Dicentric chromosome assay (DCA), 547 Dilators, 203 Diplopia (double vision), 228, 262 Direct pressure, 252 Disability, levels of, 239 Disease progression, 95 Disseminated candidiasis, 461 clinical manifestations, 461 diagnosis, 461 etiology, 461 treatment, 461 Disseminated intravascular coagulation (DIC)/purpura fulminans, 30 clinical manifestations, 467 diagnosis, 467 pathophysiology, 467 treatment, 468

Disulfiram, 980 Diuretics, 400 Diverticulitis, constipation, 371 DNA, 141 DNA-binding agents, 656-657 dexrazoxane, 656 DMSO, 656 sodium thiosulfate, 657 Dopamine agonist, 987 Doppler sonography, 416 Drainage, 308, 309 Drinking problem, 130 alcohol use disorder ED approach, 133, 134 identification and diagnosis of, 132, 133 alcohol-related cancers ICD-10 codes for, 133 potential value of ED-based alcohol SBIRT for, 135 alcohol-related disease and injury, ED encounters for alcohol use, acute conditions related to, 130, 131 alcohol use, chronic conditions related to, 131, 132 health disparities, special considerations for, 132 Drug-induced TMA, 990 Drug reaction with eosinophilia and systemic symptoms (DRESS)/ drug-induced hypersensitivity syndrome (DIHS), 452 Duloxetine, 976 Dural venous sinus thrombosis, 593 Dysphagia, 254 Dyspnea, 201, 205, 777 alleviation of dyspnea, 780 anemia, 784 arrival in ED/history, 779 benzodiazepines, 782 cardiac monitoring, 780 corticosteroids, 782 development of, 198 diagnoses, 980 EKG, 780 etiology and prevalence, 778 high-flow nasal cannula, 783 imaging, 780 laboratory studies, 780 lymphangitic carcinomatosis, 785 malignant pleural effusion, 784 neurophysiology, 778 noninvasive ventilation, 783 opioids, 781, 782 oral secretions, 785 palliative extubation, 785 ICU patients, 785 life-sustaining treatments, 785 mechanical ventilation, 785 tachypnea and prevent agitation, 786 physical exam, 779 pulmonary/respiratory system, 778 supplemental oxygen, 783 tumor burden, 785 vital sign, 780

Е

Early goal-directed therapy (EGDT) for sepsis, 399 Echocardiogram, 285 E-cigarettes, 121 Ecthyma, 460 clinical presentation, 460

diagnosis, 461 pathophysiology/etiology, 460 treatment, 461 Eczema herpeticum, 459 clinical presentation, 459-460 diagnosis, 460 pathophysiology/etiology, 460 treatment, 460 ED use at the EOL, 754 evidence strategies, 759-760 palliative care intervention, 760 palliative care programs, 759 frequency, 756, 757 future aspects, 760 healthcare providers, 755 healthcare system, 759 indicator of poor quality care, 754 literature reviews, 754 pain management, 758 patients and families, 755 system perspectives, 755 systematic review, 757 tumor and treatment factors, 758 ED visit-level data cancers with, 959 cumulative percentage of cancer patients with, 6 limitation(s), 959 method used to, 959 ED-based alcohol SBIRT, potential value of, 135 Edoxaban, 322 Effective dose, 144 Electrocautery, malignant central airway obstruction, 203, 204 Electrolyte abnormalities, 989 EKG changes, presence of, 381 hypercalcemia, 404 hyperkalemia, 403, 404 hyponatremia, 402, 403 tumor lysis syndrome, 402 Electrolyte imbalances, 362 Electronic health records (EHRs), 31, 33, 123 Electronic nicotine delivery systems (ENDS), 122 Emergency CRC resection (E-CCR), 181 Emergency department (ED), 3, 13 cervical cancer prevention in, 156 HPV vaccination, 156 screening, 157 care of oncologic patients, 5 causal loop diagram of factors, 4 determinants of, 7, 10 enabling factors, 7 international perspective on, 8 need factors, 7, 8 predisposing factors, 7 diagnosed cancer in, 94-96 distribution of cancer-related ED visits, 5 data sources, 5 disposition, 6 incidence, 5, 6 incidence and frequency, international perspective on, 6 multiple visits, 6 investigations of, 5 lung cancer, 163, 164 preventability of, 8-10 services, overutilization of, 39, 40 social worker, role of, 109, 110

staff, considerations for, 100, 101 visits, 4 Emergency department (ED) social worker CBT, 113 counseling, 113 crisis intervention, 110 language/cultural barriers, 112 oncologic patients, 110 palliative social work, 115 care conversations, 115 psychoeducation, 114 role, 109 social work initiatives health home initiative, 116 Emergency Medical Treatment and Active Labor Act (EMTALA), 38, 101 Emergency medicine, 15, 18 organizations, 171 Emergency Nursing, 983 Emergency oncologic patients, UK, 887 acute oncologic services, 887-888 assessment unit, 888 fundamental principles, 887 regional network, 888 ambulatory care, 888 cancer patients with COVID19, 889 care model, 889 COVID19, 889 management of, 887 roles, 889 Emergency Severity Index (ESI), 16, 84 tool, 16 Enasidenib, 998 Encephalopathy, 207, 209 Endocrine emergencies adrenal crisis causes, 390 diagnosis, 391 management, 391 symptoms, 390 cushing syndrome, 389 causes, 389 diagnosis, 390 management, 390 symptoms, 389 hyperthyroidism, 391 causes, 391 diagnosis, 392 management, 392 symptoms, 391, 392 myxedema coma, 392 causes, 392, 393 diagnosis, 393 management, 393 symptoms, 393 End-of-life (EOL) in advanced-stage cancer, 97, 100 care conversations ED at, 115, 116 talks, 14 Endoluminal squamous cell carcinoma, 202 Endoscopic clips, 967 Endoscopic therapy, 329 efficacy of, 328 Enemas, constipation, 373 Enteral feeding devices, issues related to, 336, 337 diarrhea, 360

Enteral nutritional supplementation, 333 Environmental ultraviolet (UV) radiation, 186 Epileptic seizures, 213 Epinephrine, 310, 977 EPIPHANY index, 319 Epiphora, 264 Epistaxis, head and neck cancer, 253 Equianalgesic dosing tables, 769 ERCP-placed stents, 334 Ervthroderma, 456 differential diagnosis, 456 MF and SS, 457 clinical manifestations, 457 diagnosis, 457 etiology, 457 treatment, 457 toxic shock syndrome, 456 clinical manifestations, 456 diagnosis, 456 pathophysiology/etiology, 456 treatment, 457 vesicles and pustules, 458 Erythroid-stimulating agents (ESAs), 548 Esophagogastroduodenoscopy (EGD), 351 Estimated GFR (eGFR), 400 EUS-guided gastroenterostomy, 349 Excess days in acute care (EDAC), 58 Expected utility theory, 26 Expected value, 25 Exposure, 143 Extraocular muscles, 262

F

Facial paresthesia, 964 Factor V, acquired deficiency of, 994 False negative rate, 27 False positive rate, 27 Family members and friends, 585 Family-witnessed resuscitation (FWR) ethical considerations, 793 family members, 792 guidelines, 793 healthcare workers, 792-793 patients, 793 Fast-Track Pathway, 16 Fatty diarrhea, 361 Febrile neutropenia antibiotic therapy, 516, 517 antifungal therapy, 517 antiviral therapy, 517 ASCO/IDSA safe discharge, 516 assessment, 515 biomarkers, 518 CISNE score, 516 definition, 513 diagnostic evaluation, 515 fungal pathogens, 514 GCSF. 517 infectious source, 514 MASCC risk index score, 516 pathophysiology, 514 risk stratification, 515 viral infections, 514 Febrile neutropenia (FN), 991 Fiberoptic tracheoscopy, 250

First time VTE, 989 Fistulas, 430, 431 Five Rights of Clinical Decision Support, 33 Flashes, 266 Flexible fiber-optic laryngoscopy, 248, 249 Floaters, 266 Fluid and electrolyte disturbances, 230 Fluoropyrimidine derivatives, 1002 5-Fluorouracil (5-FU), 273, 739, 1002 clinical presentation adverse effects, 743 cardiovascular toxicity, 744 common toxicity score, 743 death and life-threatening toxicity, 744 hand-and-foot syndrome, 744 evaluation and diagnostic testing, 744 fluoropyrimidines, 739 management of patients, 744 extracorporeal elimination, 745 G-CSF. 745 hand-and-foot syndrome, 745 pathophysiology, 743 pharmacodynamics, 740-741 cytidine deaminase, 740 deoxythymidine monophosphate, 740 metabolites, 740 RNA disruption, 741 thymidylate synthase, 740 pharmacokinetics, 741-742 toxicokinetics, 742-743 uridine triacetate, 745 Focal neurologic deficits, 217 Focused Assessment of Sonography in Trauma (FAST exam), 623 Foley balloon catheter, 337 Formal long-term strategy, 66 Formalin, 427 Fragmented measure development, 64 Frontal dysfunction with delta (FIRDA), 212 Frontal region delta activity (FIRDA) patterns, 211 Fulminant hepatic failure, 335, 336 Functional disorders, 361

G

Gamma rays, 1001 Gas embolism, 962 Gastric outlet obstruction (GOO), 348 clinical presentation and initial assessment, 349 diagnostic evaluation, 349 treatment and operative intervention, 349 Gastroduodenal stenting, for malignant obstruction, 331 Gastroenterology (GI), 327 acute pancreatitis, 333 ascites, 336 biliary obstruction, 333-335 devastating emergency, 327 enteral feeding devices, issues related to, 336, 337 gastrointestinal hemorrhage bleeding related to tumors, 328-330 Mallory-Weiss tear, 330 radiation proctitis, 330, 331 hepatic decompensation, fulminant hepatic failure, 335, 336 luminal obstruction, 331, 332 oncologic emergencies, 328 Gastrointestinal bleeding, 328, 329, 349 clinical presentation and initial assessment, 350

diagnostic evaluation, 350, 351 treatment and operative intervention, 351 Gastrointestinal emergencies gastrointestinal perforation and fistula, 430, 431 malignant bowel obstruction, 429, 430 causes of, 429 diagnosis of, 430 laboratory studies, 430 management, 430 plain abdominal radiography, 430 Gastrointestinal epithelium, 352 Gastrointestinal hemorrhage bleeding related to tumors, 328-330 Mallory-Weiss tear, 330 radiation proctitis, 330, 331 Gastrointestinal mucositis, 977 Gastrointestinal (GI) perforation, 430, 431 Glioblastoma, intratumoral hemorrhage in, 216 Global Alliance for Vaccination and Immunization (GAVI), 154 Glomerular filtration rate (GFR), 400 Graft versus host disease (GVHD), 93, 352, 364, 503 complications, 95 GVHD -induced diarrhea, 359 Granulocyte colony stimulating factor (G-CSF), 527 Granulocyte-macrophage colony-stimulating factor (GM-CSF), 548 Graves disease, 391 Gross hematuria, 408-411 Gynecologic cancer, 426 genitourinary emergencies in urinary tract obstruction, 431, 432 vaginal cuff dehiscence, 431 surgical emergencies, acute blood loss, 426, 427 Gynecologic Oncologic Global Curriculum and Mentorship Program, 156 Gynecology, 426 acute blood loss, 426 hemorrhagic cystitis treatment, 427, 428 invasive bleeding management, 427 management, 427 non-invasive bleeding management, 427 in cancer urinary tract obstruction, 431, 432 vaginal cuff dehiscence, 431 gastrointestinal emergencies, gynecologic cancer patients gastrointestinal perforation and fistula, 430, 431 malignant bowel obstruction, 429, 430 pain, 432, 433 pelvic infections, 428 antibiotic therapy, 429 laboratory findings, 429 management, 429 Gynecomastia, 415

H

Hand-and-foot syndrome, 743 Hartmann's resection, 347 Head and neck cancer, 247 acute airway, potential for, 248 airway management, 248 SARS-CoV-2 pandemic, considerations during, 250, 251 secured airway, 249, 250 unsecured airway, 248, 249 bleeding management, 251 acute arterial bleeding, 251, 252 carotid blowout, 252

epistaxis, 253 internal jugular vein bleeding, 252 tracheoinnominate fistula, 252, 253 examination, 247 neck abscess versus occult SCCHN cystic cervical lymph node metastasis, 255 oropharyngeal squamous cell carcinoma, 247 oropharynx, ear infection versus occult SCCHN of, 255, 256 sinusitis versus occult sinonasal malignancy, 255 treatment complications, management of chemotherapy, 254, 255 radiation therapy, 254 surgical complications, 253, 254 Headache, pituitary tumor apoplexy, 227, 228 Healthcare disparities breast cancer, 885 colorectal cancer, 884 cultural humility, 881 definitions, 881-882 explicit bias, 881 schizophrenia, 884 sexual minorities, 882, 883 transgender patients, 882 vulnerable populations, 882 Health economics, lung cancer, 170 Health Insurance Portability and Accountability Act (HIPAA), 31, 32, 959, 985 Health professionals, 192 Health Services Research (HSR), 79 Health services/resource utilization, patient navigation, 78, 79 Heavy alcohol use, 130-132 Hematochezia, 986 Hematoma of upper neck, 254 Hematopoietic cell transplantation (HCT), 699 acute complications after allogenic, 703 diffuse alveolar hemorrhage, 705 idiopathic pneumonia syndrome, 704 renal complications, 705 risk of infection, 703 VOD treatment, 703 acute graft-versus-host disease, 706 clinical manifestations, 706 organ staging and grading, 706 acute infections, 704 acute leukemias, 699 acute neurologic complications, 705-706 apheresis compliactions, 700 autologous GVHD, 703 DMSO toxicity, 700 engraftment syndrome, 702 gastrointestinal complications, 701-702 G-CSF administration, 699 ischemic complications, 700 splenic rupture, 700 graft failure, 707 GVHD prophylaxis, 699 myelodysplastic syndrome, 699 neurologic complications, 705 oropharyngeal mucositis, 701 respiratory failure, 704 TMA diagnosis, 705 transfusion-associated GVHD, 703 Hematopoietic stem cell transplantation (HSCT), 502-503, 973 and AKI, 401, 402 Hematuria, 407, 421 catheters, 409

Hemipelvectomy, 446 Hemoptysis, 200, 201, 205, 309 etiologies, 309, 310 massive hemoptysis, 310 Hemorrhage, 226, 229, 426, 963 Hemorrhagic cystitis (HC), 426, 428-429 management of, 428 treatment, 427, 428 Hemorrhagic infarction of pituitary tumor, 226 Hemostatic radiotherapy (RT), 427 Heparin-induced thrombocytopenia (HIT), 466 diagnosis, 467 pathophysiology, 467 treatment, 467 Hepatic decompensation, fulminant hepatic failure, 335, 336 Hepatocellular carcinoma (HCC), 131, 133 Hestia criteria, 320 Heterochromia, 266 Heterogeneity, 225 Heuristics, 26 High-flow nasal cannula (HFNC) oxygen therapy, 304 High-grade squamous intraepithelial lesions (HSIL), 152 Highmark Foundation, 79 High-risk HPV (hrHPV) type, 152 HIPAA, see Health Insurance Portability and Accountability Act (HIPAA) Hodgkin lymphoma, 291 Hormonal deficit, 987 Horner syndrome, 266 Hospice in ED, 821 case study, 825-828 discharge home without services, 823 hospice referral process, 823 level of care, 825 patient and support reasons, 822 patient eligiblity, 823 patient's clinical status and wishes, 823 patients enrollment, 822 prognostic understanding, 823 referral process, 824 Hospice reimbursement model, 44, 45 Hounsfield unit, 996 HPV, 961 testing, 155 vaccination challenges and barriers, 154 HPV vaccination, 152 HPV vaccine protection, 960 HTN, 274 Human leukocyte antigen (HLA), 546 Human papillomavirus (HPV), 152, 961 testing, 155 vaccination, 152 challenges and barriers, 154 protection, 960 Human tetanus immunoglobulin (HTIG), 441 Hyaluronidase, 997 Hydronephrosis, 414, 415 Hyperbaric oxygen therapy (HBOT), 331 Hypercalcemia, 969 causes, 383 diagnosis, 384 electrolyte abnormalities, 404 management, 384 symptoms, 383 Hypercalcemia of malignancy, constipation, 371, 372 Hypercapnic acidosis, 965 Hyperglycemia

causes, 386, 387 diagnosis, 387 management, 387 symptoms, 387 Hyperkalemia, 380 causes, 380 diagnosis, 380 electrolyte abnormalities, 403, 404 management, 380, 381 symptoms, 380 Hyperleukocytosis, 993 complications, 482 incidence of, 483 white blood cell, 482 Hypermagnesemia causes, 382 diagnosis, 382 management, 382 symptoms, 382 Hypernatremia, 377 causes, 378 diagnosis, 378 management, 378 symptoms, 378 Hyperosmolar hyperglycemic nonketotic coma, 387 Hyperphosphatemia causes, 385 diagnosis, 385 management, 385 symptoms, 385 Hypertension (HTN), 274, 972 clinical consequences of, 274 diagnosis, 275 mechanisms, 275 treatment, 275 Hyperthermic intraperitoneal chemotherapy (HIPEC), 803 Hyperthyroidism, 391 causes, 391 diagnosis, 392 management, 392 symptoms, 391, 392 Hyperuricemia, 389 Hyperventilation, 211 Hypoactive delirium, 995 Hypocalcemia, 384 causes, 384 management, 385 symptoms, 384, 385 Hypoglycemia causes, 387, 388 diagnosis, 388 management, 388 symptoms, 388 Hypokalemia, 381 causes, 381, 382 diagnosis, 382 management, 382 symptoms, 382 Hypomagnesemia, 382 causes, 383 diagnosis, 383 management, 383 symptoms, 383 Hyponatremia, 378 algorithm, evaluation and management of, 379 causes, 378 diagnosis, 379

electrolyte abnormalities, 402, 403 management, 379 symptoms, 379 Hypophosphatemia causes, 385, 386 diagnosis, 386 management, 386 symptoms, 386 Hypotension, 989 Hypothyroidism, constipation, 371

I

Iatrogenic pneumothorax, 307 Ibrutinib, 282 ICI therapy, 988 Ifsosfamide, 968 toxicity, 214 Image-guided percutaneous therapies, 308 Immediate Jeopardy (IJ), 102 Immune checkpoint inhibitor-induced cardiomyopathy, 277, 279 Immune checkpoint inhibitor therapy, diarrhea, 359 Immune effector cell-associated neurotoxicity syndrome, 732 management, 732 signs, 732 Immune-effector cell-associated neurological syndrome (ICANS), 208 Immune-mediated colitis, 364 Immune-related adverse events (irAEs), 726 cardiac irAEs, 726 emergency physicians, 726 endocrine irAEs, 726 gastrointestinal irAEs, 729 HLH, 729 hypophysitis, 728 immunosuppression, 726 integumental irAEs, 729 neurologic irAEs, 729 pulmonary irAEs, 729 rheumatic manifestations, 730 thyroiditis, 728 Immune-related bullous pemphigoid-like (irBP) eruption, 458 clinical manifestations, 458 diagnosis, 458 pathophysiology/etiology, 458 treatment, 458 Immunotherapy AKI and, 400 myxedema coma, 393 Immunotherapy-induced pericarditis, 276 Impending pathologic fractures, orthopedics, 440 Incidentaloma, 27 Independence at Home (IAH), 116 Indwelling pleural catheters (IPCs), 309 Infections, 94, 358 nursing, 94 orthopedics, 442 Infectious colitis, 361 Infectious disease antimicrobial prophylaxis, 502, 503 central nervous system, 508 chemotheraphy for solid organ tumors, 504 cytotoxic chemotherapy, 504 diagnostic work up, 505 febrile neutropenia, 501, 505 future issues and considerations, 509 gastrointestinal syndromes, 508 GVHD treatment, 503

healthcare-associated infections, 508 history and physical examination, 505 HSCT, 502, 503 immune defects, 502 immune dysfunction, 501 MASCC risk assessment, 506 oncologic patient in ED, 504 oncologic patient with fever, 505 pulmonary syndromes, 507 skin and soft tissue infections, 508 T-cell-targeting therapies, 501 treatment and management, 506-507 Infectious enteritis, 359, 360 Infectious keratitis, 263 Inferior vena cava tumor thrombus, 412 Inflammation of iris, 264 Inflammatory bowel disease (IBD), 143 Inflammatory interstitial pneumonitis syndrome, 311 Information systems (IS), 33 Information technology systems Health Insurance Portability and Accountability Act, 32 telehealth, 30, 31 In-hospital CPR, cause of, 980 Initial chemotherapy infusion, 964 Inpatient management plan, 583 Inpatient quality reporting (IQR), 46 Inpatient unit care, 981, 982 Institute for Safe Medication Practices (ISMP), 32 Interdisciplinary collaboration, nursing, 102, 103 Intermittent hemodialysis (IHD), 399 Internal jugular vein bleeding, head and neck cancer, 252 International normalized ratio (INR), 335 Interventional radiology (IR), 329 Intracranial hemorrhage (ICH) diagnosis, 216 importance and causes, 215, 216 presentation, 216 treatment and prognosis, 216 Intracranial mass effect, 591 Intracranial pressure (ICP), 209 Intratumoral hemorrhage, 228 in glioblastoma, 216 Intravascular therapy, superior vena cava syndrome, 299, 300 Intrinsic renal disease, 398 Intubation, 961, 972 Iohexol clearance, 400 Ionizing radiation, 141 clinical decision support appropriateness criteria, 146 clinical decision rules, 146 computerized interventions, 146, 147 quality metrics and regulatory efforts, 147 shared decision-making with patients, 147 cumulative radiation, contribution to, 139, 140 increased utilization, reasons for, 140, 141 repeat/multiple CT imaging, patterns of, 141 transformative tool, medicine, 139 utilization, explosion, 139 deterministic vs. stochastic effects of, 141 dose estimates, 144

low-dose ionizing radiation and cancer, 142, 143

measures of, 143-145 and optimize imaging provider and patient awareness, 146 technological improvements, 144, 145 risk, 143

CT

irAEs, *see* Immune-related adverse events (irAEs) Ischemic bowel, 425 Ischemic cardiovascular, 269–272 arterial thrombosis, 272 coronary artery vasospasm, 273 peripheral artery disease, 273, 274 premature coronary artery disease, 272, 273 vasculitis, 273

J

JAK2, 994 James Diagnostic Clinic, 19

K

Karnofsky performance score (KPS), 242, 430 Kidneys, 397 Kyphoplasty, 242

L

Language/cultural barriers, palliative social work, 112 Large bowel obstruction (LBO), acute abdomen, 346 clinical presentation and initial assessment, 347 diagnostic evaluation, 347 treatment and operative intervention, 347, 348 Laryngeal cancer, 987 Laryngeal mask anesthesia (LMA), 248 Laryngectomy stoma, 250 Lasers, malignant central airway obstruction, 203 1-Asparaginase, 318, 393, 994 Lay navigator, patient navigation, 77 LCS barriers, proposed solutions to, 168, 169 LDCT screening, barriers to, 166, 168 Learning healthcare system, HIT, 67 Leg swelling, 420 Length of stay, 15, 16 Lentigo maligna, 188 Leptomeningeal disease, 260 Leukemia, 14, 95, 974 Leukemias, see leukostasis Leukostasis disposition and prognosis, 487-488 ED evaluation, 484-486 ED management, 486 airway/breathing, 487 circulation, 486 definitive treatment, 487 emergency physicians, 487 epidemiology, 482 future research, 488 health economics, 483 pathophysiology, 483 in ALL, 484 in AML, 484 in CLL, 484 in CML, 483 hematopoiesis, 483 myelodysplastic syndrome, 484 presentation, 484 treatment costs, 483 Limitations of published oncologic emergency research, 916-927 Lobar bronchi distal, patency of, 986 Low-and middle- income countries (LMICs), 151 cervical cancer, prevention, 156

Low-dose ionizing radiation, and cancer, 142, 143 Low molecular weight heparin (LMWH), 319 Low tidal volume ventilation (LTVV), 304, 305 Lower abdominal pain, 407 Lower gastrointestinal bleeding (LGIB), 349 Lubiprostone, 373 Lubricants, constipation, 373 Luer-lock port, 410 Luminal obstruction, 331, 332 Lung cancer (LC), 161 cancer prevention and screening, 170, 171 cigarettes and smoke exposure, 162 ED imaging and, 163, 164 environmental risk factors, 162 epidemiology and high-risk populations, 162, 163 health economics, 170 in Emergency Department, 163 LCS barriers, proposed solutions to, 168, 169 LDCT screening, barriers to, 166, 168 National Lung Screening Trial, 165, 166 Nederlands-Leuvens Longkanker Screenings Onderzoek Trial, 169 pathology, prognosis and treatment, 163 prevention, 164 prognosis for, 161 screening with chest X-ray and overdiagnosis problem, 164, 165 computed tomography for, 165 future of, 169, 170 Lung cancer screening (LCS), 161 in ED. 171 online resources for, 171 shared decision making, 171 Lung CT Screening Reporting & Data System (Lung-RADS®), 167 Lymphangitic carcinomatosis (LC), 785 Lymphomas, 291, 404, 980, 984

М

Maculopapular drug rash, 992 Maculopapular eruptions, 451, 452 drug eruptions diagnosis, 452 pathophysiology/etiology, 452 **GVHD**, 453 clinical manifestations, 453-454 diagnosis, 454 pathophysiology/etiology, 454 treatment, 454 immune-related maculopapular rash clinical manifestations, 452-453 diagnosis, 453 pathophysiology/etiology, 453 treatment, 453 viral exanthems, 453 clinical manifestations, 453 diagnosis, 453 pathophysiology/etiology, 453 treatment, 453 Maculopapular eruptions drug eruption, 452 clinical manifestations, 452 treatment, 452 Magnesium, 382 Magnetic resonance angiography (MRA), 987 superior vena cava syndrome, 296 Malignant bowel obstruction (MBO), 342 gastrointestinal emergencies, 429

causes of, 429 diagnosis of, 430 laboratory studies, 430 management, 430 plain abdominal radiography, 430 Malignant central airway obstruction (MCAO), 197 classification, 198 cold tools airway stenting, 204, 205 cryoprobe and cryospray, 204 development of symptoms, 199 epidemiology, 197, 198 evaluation, 199 management, 199-202 mechanical debulking, 203 dilators, 203 microdebrider, 203 rigid bronchoscopy, 203 presentation of, 198, 199 therapeutic bronchoscopy, 202 thermal tools, 203 argon plasma coagulation, 204 electrocautery, 203, 204 lasers, 203 Malignant obstructions, 331 Malignant small bowel obstruction (MaSBO), 346 Malignant spinal cord compression (MSCC), 238 constipation, 371 epidemiology, 238 history, 238, 239 imaging, 239, 240 management, 240 medical therapy, 241 nursing efforts, 240, 241 pathophysiology, 238 physical examination, 239 prevention, 243 prognosis, 243 radiotherapy, 242 steroids, 242 surgical therapy, 241, 242 surveillance and diagnostic practices, 242 Mallory-Weiss tear, 330 Manual irrigation, 409 Massive hemoptysis, 310, 975 McGill pain questionnaire, 766 MD Anderson and Memorial Sloan Kettering Cancer Center, 14 Mechanical debulking, 203 dilators, 203 microdebrider, 203 rigid bronchoscopy, 203 Medicare Access and CHIP Reauthorization Act of 2015 (MACRA), 46 Medicare Care Choices Model, 45 Medicare Hospice Benefit, 44 Medicare Prescription Drug, Improvement, and Modernization Act of 2003 (MMA), 46 Medication-assisted treatment (MAT), 582 buprenorphine, 582 naltrexone, 582 Melanoma, 185 Breslow thickness and 10-year survival, 190 clinical manifestations, 186-188 diagnosis, 190 epidemiology, 186 investigation and diagnosis, 189

mortality, predictor of, 960 pathophysiology, 186 prevention and screening, 186 primary prevention, 190, 192 risk factors, 186 secondary prevention, 192 subtypes, 187 teledermatology, 189 treatment, 189, 190 Memorial Hermann Health System, 79 Memorial pain assessment card, 766 Memorial Sloan Kettering Cancer Center, 16, 17 Mental status, 207 Merit-based Incentive Payment System (MIPS), 46 Mesenteric angiography, 351 Metabolic emergencies hypercalcemia causes, 383 diagnosis, 384 management, 384 symptoms, 383 hyperglycemia causes, 386, 387 diagnosis, 387 management, 387 symptoms, 387 hyperkalemia, 380 causes, 380 diagnosis, 380 management, 380, 381 symptoms, 380 hypermagnesemia causes, 382 diagnosis, 382 management, 382 symptoms, 382 hypernatremia, 377 causes, 378 diagnosis, 378 management, 378 symptoms, 378 hyperphosphatemia causes, 385 diagnosis, 385 management, 385 symptoms, 385 hypocalcemia, 384 causes, 384 management, 385 symptoms, 384, 385 hypoglycemia causes, 387, 388 diagnosis, 388 management, 388 symptoms, 388 hypokalemia, 381 causes, 381, 382 diagnosis, 382 management, 382 symptoms, 382 hypomagnesemia, 382 causes, 383 diagnosis, 383 management, 383 symptoms, 383 hyponatremia, 378

Metabolic emergencies (cont.) causes, 378 diagnosis, 379 management, 379 symptoms, 379 hypophosphatemia causes, 385, 386 diagnosis, 386 management, 386 symptoms, 386 tumor lysis syndrome, 388 causes, 388, 389 diagnosis, 389 management, 389 symptoms, 389 Metastatic B-cell lymphoma, 211 Metastatic breast cancer, 973 Metastatic cholangiocarcinoma, 973 Metastatic disease, 970 Metastatic pancreatic cancer, 982 Metastatic renal cell carcinoma, 439, 440 Metastatic squamous cell carcinoma, 964 Methadone, 979, 983, 1001 Methadone maintenance therapy (MMT), 582 Methotrexate (MTX), 399, 977 Methylnaltrexone, 373, 990 Methylprednisolone, 281 Metoclopramide, 990 Metoprolol, 377 Metronidazole, 357 Michigan Oncologic Medical Home Demonstration Project, 77 Microdebrider, 203 Microhematuria, 407, 409 Microsimulation Screening Analysis Lung (MISCAN-Lung) model, 170 Minimally invasive adenocarcinoma (MIA), 166 Minimally invasive procedures, 242 Mini-mental state examination (MMSE), 208 Minority stress model, 982 Mirels' score, 440 Models of care acute care facilities, 14 Cancer Emergency Department, 19-21 comparative survival data, 13 factors, 14 issues, 15 life-threatening illness, 14 Memorial Sloan Kettering Cancer Center, 16, 17 Ohio State University Medical Center, 17-19 specialized care, 13 University of Texas MD Anderson Cancer Center, 15 Moh's paste, 427 Monoclonal antibodies, diastolic and restrictive cardiomyopathy, 279 Monsel solution(ferric subsulfate), 427 Motivational interviewing, 985 Mucormycosis, 261 Mucositis, 679, 977 anatomy, 680 assessment tools, 682 diagnosis, 681-682 disposition/follow-up, 685-686 documentation, 686 dose-limiting toxicity, 679 epidemiology, 681 future vision, 686 gastrointestinal mucositis, 679

amifostine, 685

antihistamines, 685 antimicrobials, 685 dietary modifications, 685 formalin, 685 hyperbaric oxygen therapy, 685 nutritional supplements, 685 octreotide, 685 palifermin, 685 sucralfate, 685 health economics, 681 health services, 686 history, 680 oral mucositis, 679 aloe vera, 684 anti-inflammatory agent, 682 curcumin, 684 doxepin hydrochloride, 684 elemental diet formulations, 684 gabapentin, 684 glutamine, 684 honey, 684 opioid analgesics, 684 oral care, 684 oral cryotherapy, 682 palifermin, 683 photobiomodulation, 683 vitamin E, 684 zinc, 684 pathophysiology, 680 pitfalls, 686 presentation, 681 toxic mucositis, 681 treatment complications, 686 Multilobar infiltrates, 973 Multiphase Optimization Strategy (MOST), 124 Multiple myeloma and AKI, 401 Multiple sclerosis, 260 Musculoskeletal emergencies appendicular skeleton, 605 hypercalcemia, 608 impending fractures, 606 pain, 607 pathologic fractures, 605 vertebral fractures, 606 Myeloproliferative neoplasms antiplatelet therapy, 495 antithrombotic therapy, 495 budd-chiari syndrome, 494 platelet reduction, 495 Myocarditis, 280 diagnosis, 281 presentation, 280, 281 treatment, 281 Myxedema coma, 392 causes, 392, 393 diagnosis, 393 management, 393 symptoms, 393

Ν

Naldemedine, 373 Naloxegol, 373 Narcotic bowel syndrome (NBS), 368, 369, 372, 374 diagnostic criteria for, 369 Nasolacrimal duct obstruction, 264 Nasotracheal intubation, 248 National Cancer Act, 34, 76 National Cancer Institute (NCI), 76 National Cancer Institute Common Toxicity Criteria, 360 National Comprehensive Cancer Network (NCCN), 43 National Healthcare Safety Network (NHSN), 55 National Health Interview Survey (NHIS), 122 National institutes of health (NIH), 915 National Lung Screening Trial (NLST), 165, 166 National Program of Cancer Registries (NPCR), 34 National Quality Forum (NQF), 46 National Survey on Drug Use and Health (NSDUH), 130 Nationwide Emergency Department Sample (NEDS), 163 Navigator program, 983 NBS, see Narcotic bowel syndrome (NBS) Nebulized tranexamic acid, 965, 988 Neck abscess, 255 Neck breathers, 250 Neck mass, 988 Necrotizing enterocolitis, 991 Necrotizing fasciitis, 455, 970 orthopedics, 442, 443 Nederlands-Leuvens Longkanker Screenings Onderzoek (NELSON) Trial, 169 Needle aspiration (NA), 308 Negative predictive value (NPV), 27 Neodymium-yttrium-aluminum-garnet (Nd:YAG) laser, 203, 328 Neoplasia, 328 Nephrectomy abdominal collections, 421 hematuria, 421 pulmonary symptoms, 421 Nephrostomy tubes, 415 Neuroendocrine cancer, 990 Neurologic emergencies, 591-595 complications, 595 dural venous sinus thrombosis, 592 edema, 591 hydrocephalus, 592 intracranial hemorrhage, 592 intracranial mass effect, 591 leptomeningeal metastatic disease, 592 spinal pathology, 594 stroke, 593 Neutropenic enterocolitis (NE), 351, 352, 364, 521 antibiotics for empiric treatment, 525 clinical presentation and initial assessment, 352, 522 description, 522 diagnostic criteria, 352, 522, 523 diagnostic imaging, 524 granulocyte colony stimulating factor, 527 initial workup, 524 medical versus. surgical treatment, 525 pathophysiology, 523 pediatric considerations, 524 risk factors, 523 signs and symptoms, 522 surgical intervention, 526 treatment and operative intervention, 352 Neutropenic enterocolitis (NEC), 352 Neutropenic fever, 18, 990 Neutropenic ileitis, 362 Neutrophil extra-cellular traps (NETs), 272 Nicotine replacement therapy (NRT), 124, 960 Nicotine-containing solution, 121 N-methyl d-aspartate receptor, 369 No-Duty-to-Treat Principle, 38, 39 Nonavalent vaccine, 154

Nonconvulsive convulsive status epilepticus (NCSE), 212, 214 Nonfunctional pituitary macroadenoma, 229 Noninvasive positive-pressure ventilation (NIPPV), 304 Noninvasive ventilation, 999 Non-ionizing radiation, 141 Non-ischemic vascular conditions hypertension, 274, 275 venous thromboembolism, 275, 276 Nonobstructive hydronephrosis, 414, 415 Non-opioid analgesics acetaminophen, 768 NSAIDs COX, 768 nonselective/selective, 768 side effects, 768 Nonsteroidal anti-inflammatory medications (NSAIDs), 768 Non-thrombotic pulmonary embolism (NTPE), 313 Normal saline, 969 Novel antineoplastic therapies CAR-T cell therapy, ICANS, 732 CAR-T-cell therapy, 730 cytokine release syndrome, 730 differentiation syndrome, 724 IDH inhibitors, 724 uric acid, 724 infusion-related reaction, 724 irAEs, 726 cardiac irAEs, 726 emergency physicians, 726 endocrine irAEs, 726 HLH, 729 immunosuppression, 726 integumental irAEs, 729 neurologic irAEs, 729 pulmonary irAEs, 729 rheumatologic irAEs, 730 QT prolongation, 725 sinusoidal occlusion syndrome, 725 clinical presentation, 725 emergency physicians, 726 treatment, 726 tumor lysis syndrome, 726 Nuclear power plant (NPP) incidents, 545 Numerical rating scales (NRS), 766 Nursing abdominal symptoms, 89, 93 advanced-stage cancer, end-of-life in, 97 altered mental status, 87, 88 back pain, 89 chest pain and shortness of breath, 84, 87 COVID 19, 99 preventing cancer patients from, 100 screening and safety measures, 100 ED staff considerations for, 100, 101 diagnosed cancer in, 94, 95 emergency nursing and oncologic emergencies, 83 infection, 94 interdisciplinary collaboration, 102, 103 malignancy progression, antineoplastic treatments, and general medical emergencies, 95-97 oncologic emergencies general assessment, 84 triage, 84 oncologic emergency departments, regulatory standards for, 101, 102 post patient screening and treatment room assignment, 101 SARS CoV 2, 97

0 Obstipation, 370 Occluded biliary stents, 335 Occult fecal bleeding, 986 Occult SCCHN cystic cervical lymph node metastasis, 255 Occult sinonasal malignancy, 255 Ocular graft-versus-host disease (GVHD), 263 Ohio State University Medical Center (OSUWMC), 17-19 Ohio State University Wexner Medical Center, 14 OIC, see Opioid-induced constipation (OIC) Oligoanalgesia, 767 Oncologic emergencies, 13, 15, 19, 20, 296, 589 CT technology, 589 emergency nursing and, 83 fluoroscopy, 590 general assessment, 84 imaging modalities, 589 interventional radiology, 590 magnetic resonance imaging, 590 nuclear medicine, 590 plain radiography, 589 radiologic evaluation, 590-608 telehealth, 31 triage, 84 ultrasonography, 590 Oncologic emergency care episode of, 64 health policy for, 66, 67 national quality measurement, vision for, 65, 66 state of national quality measurement, 65 Oncologic emergency medicine, 67, 68 Oncologic Emergency Nursing (OEN), 83 Oncologic care, 14 Oncologic emergency departments models, 14 regulatory standards for, 101, 102 Oncologic nurse navigator (ONN), 76, 77 Oncologic patient's caregiver, OSW connecting an, 116 Oncologic Pod, 18, 19 Oncologic social worker (OSW), 77, 115 Open fractures, orthopedics, 441 Open-lung ventilation, 305 Ophthalmology, 259 acute visual loss, 259-262 diplopia, 262, 263 emergencies, 259 epiphora, 264 flashes and floater, 266 proptosis, 265 ptosis, 265, 266 red eye, 263, 264 Ophthalmoplegia, 234 Opioid analgesics, 998 cross-tolerance, 769 equianalgesic dose equivalent, 769 equianalgesic dosing tables, 769 fentanyl, 769 intramuscular route, 768 intravenous route, 768 methadone, 769 oral route, 768 receptor, 768 semisynthetic/synthetic, 768 side effects, 770 constipation, 770 nausea, 770

pruritus, 769 respiratory depression, 770 subcutaneous/rectal route, 768 tolerance, 769 Opioid use disorder (OUD), 59, 580 Opioid-induced constipation (OIC), 368, 373 Opioids, 999 constipation, 369 Opportunistic fungal infections, 466 clinical manifestations, 466 diagnosis, 466 pathophysiology/etiology, 466 treatment, 466 Optic neuritis, 260 Oral antibiotics, 995 Oral cancer, history of, 218 Oral dosing, 742 Orbital cellulitis, 265 Orbital lymphoma, 265 Order Set, 32, 33, 959 Oropharynx, ear infection versus occult SCCHN of, 255, 256 Orthopedics, 437 abscess, bursitis, tenosynovitis, 443 arthroplasty, 445 cellulitis, 442 compartment syndrome, 441, 442 impending pathologic fractures, 440 infections, 442 necrotizing fasciitis, 442, 443 open fractures, 441 osteomyelitis, 444 pathologic fractures, 437-440 pelvis, 446 periprosthetic fractures, 445 prosthetic dislocation, 445, 446 septic arthritis, 443, 444 trauma, 437 Orthostatic hypotension, 357 Osmotic diarrhea, 361 Osmotic laxatives, constipation, 373 Osteomyelitis, orthopedics, 444 OSUWMC ED triage, 19 Outpatient management plan, 584 Outpatient oncologic social worker, 112, 113 Outreach and Chronic Disease Prevention Act, 76 Ovarian cancer, 240, 425 Overdiagnosis, 27 Overutilization, 140, 146 Oxidative stress, 275

P

Paclitaxel, 430 Pain, 341 breakthrough pain, 765 consultation, 773 EPECTM-EM, 772 palliative sedation, 773 rapid titration, 772 gynecologic, 432, 433 nociceptive somatic pain, 766 visceral pain, 766 non-cancer-related pain syndromes, 766 oligoanalgesia, 767 opioid safety, 771

pain emergency, 766 palliative care in emergency medicine project, 773 prevalence in cancer, 765 severity assessment tools, 766 treatment non-opioid analgesics, 768 WHO analgesic ladder, 767, 768 Palliative care, 809-810, 981, 1000 advanced integration programs, 812 assessment at time of admission, 815 assessment for ED, 815 basic integration programs, 811 consults, 814 ED-focused advanced integration programs, 812 generalist versus specialist levels, 810-811 health economics, 816 hospice collaborations, 815 inpatient units, 814 institutional and community resources, 814 integrated emergency medicine, 811 integration initiative, 812-815 literature and resources, 813 monitoring integrated initiatives, 815-816 principles and elements, 810 proposed goals, 810 recruit team members, 813 superior vena cava syndrome, 300, 301 traditional consultation programs, 811 Palliative care, inadequate access to, 43, 44 Palliative care (PC) in ED, 843 advanced integration, 855 attitudinal studies, 851-852 educational needs, 852 emergency physicians, 851 end-of-life care, 852 healthcare workers, 851, 852 specialist clinicians, 852 basic integration, 854 clinicians education, 857-858 death disclosure, 857 EPEC-EM, 857 training efforts, 857-858 communication, 855-856 descriptive studies, 843 access and coordination of care, 851 in cancer population, 850 general descriptive, 844 pediatric, 851 ED-focused advanced integration, 855 end-of-life care, 855 outcomes research, 854 research, 845-850 resource allocation, 856-857 after ED, 857 community-based, 856 within ED, 856-857 screening, 852 FAST criteria, 854 NCCN guidelines, 854 P-CaRES, 853 SPEED tool, 853 surprise question, 853-854 systematic review, 853 traditional model of consultation, 854 Palliative radiation therapy, 433 Palliative sedation, 772

Palliative social work, 109, 115 cancer patients, psychosocial issues, diagnosis, 110, 111 caregiver distress, 112 clinical interventions, 113 cognitive behavioral therapy, 113 complex case management, 110 crisis intervention, 110, 113, 114 ED communication in, 111 initiative in. 115 primary healthcare site, 111, 112 social worker, role of, 109, 110 end of life, care conversations ED at, 115, 116 health home initiative, 116, 117 interventions, 114, 115 language/cultural barriers, 112 with oncologic patients, 110 oncologic patient's caregiver, OSW connecting an, 116 outpatient oncologic social worker, 112, 113 psychoeducation, 114 relaxation techniques, 113 role of, 115 supportive counseling, 113 worker, 115 Palliative surgery, 797 abdominal pain celiac plexus involvement, 804 multiple myeloma, 804 neutropenia, 804 anorectal infections, 803 bowel perforation, 802 gastrointestinal bleeding, 801 endoscopy, 801 gastric hemorrhage, 801 laboratory analysis, 801 small bowel tumors, 801 gastrointestinal obstruction, 798 emesis and abdominal distention, 799 gastric outlet obstruction, 799 imaging, 798 laboratory analysis, 798 large bowel obstruction, 800 malignant bowel obstruction, 798, 799 objective assessment, 798 small bowel obstruction, 800 treatment, 798 venting gastrostomy tubes, 800 malignant ascites, 803 HIPEC, 803 paracentesis, 803 morbidity and mortality rates, 798 obstructive jaundice, 802 outcome measures, 804 patient selection, 798 wound complications and infections, 801 Pancoast tumor, 266 Pancreatic adenocarcinoma, 334 Pancreatic cancer, 967 Pancreatic toxicity, 359 Pancreatitis, 328 Panitumumab, 966 Pap test, 155 Papilledema, 211 Paracentesis, 17, 20 Paraneoplastic diarrhea, 364 Paraneoplastic glomerular disease, 402

Paraneoplastic pemphigus (PNP), 464 clinical manifestations, 464 diagnosis, 464 pathophysiology, 464 treatment, 464 Parathyroid hormone-related protein (PTHrP)-mediated hypercalcemia, 383 Parenteral nutrition (PN), 430 Paroxysmal nocturnal hemoglobinuria (PNH), 495 Pathologic fractures, orthopedics, 437-440 Patient-centered medical homes (PCMH), 45 patient navigation, 77, 78 Patient dissatisfaction, 41 Patient education, 243 Patient navigation, 75, 79 health services/resource utilization, 78, 79 management, 76 lay navigator, 77 oncologic nurse navigator, 76, 77 oncologic social workers, 77 patient centered medical home, 77, 78 programs, 76 Patient Navigation Research Program (PNRP), 78 Patient Protection and Affordable Care Act, 38 Peaked T wave, 380 Pediatric cancer, 471 anatomy, 472 chemotherapeutic agents, 474 common pitfalls, 477 complications, 477 disposition/follow- up, 476 epidemiology, 472 evaluation abdominal radiographs, 475 diffuse alveolar hemorrhage, 475 oncologic processes, 475 pancreatitis patient, 475 tumor lysis syndrome, 475 future needs/vision, 477 health economics, 472-473 health services, 477 history and background, 472 management coagulopathy, 476 cytomegalovirus, 476 electrolyte repletion, 476 fever, 476 hyperleukocytosis, 475 mediastinal mass patients, 476 neuroblastoma patients, 476 pancreatitis, 476 pericardial effusion, 476 medicolegal, 477 pathophysiology hyperleukocytosis, 472 tumor lysis, 472 undergoing therapy, 472 presentation/diagnosis, 473 anemia, 473 blood dyscrasias, 473 chemotherapeutic agents, 474 graft-versus-host disease, 474 hemoptysis, 474 hypercoagulopathic side effect, 473 in gastrointestinal tract, 474 metastatic brain lesions, 473

pancreatitis, 474 renal system injuries, 474 prevention/upstream drivers, 477 prognosis/treatment, 476 quality indicators, 477 symptoms, 473 Pediatric cancers, 971 Pediatric Symptom Checklist (PSC), 54 Pelvic abscess, 428 Pelvic infections, 428 antibiotic therapy, 429 laboratory findings, 429 management, 429 Pelvic osteosarcoma, 447 Pelvis, orthopedics, 446 Pembrolizumab, 377 Pembrolizumab-associated myopathy, 219 Penile cancers, 417, 418 Pentoxifylline, 312 Percutaneous catheters, 335 Percutaneous nephrostomy (PCN) tube, 399 Percutaneous transhepatic cholangiography (PTC), 334 Perforation, colorectal cancer, 183 Perform pleuroscopy, 988 Pericardial diseases, 276 pericardial effusion, 277 pericarditis, 276, 277 Pericardial effusion, 277 Pericardiocentesis, 976 Peripheral acting u-opioid receptor antagonists (PAMORAs), 372, 373 Peripheral artery disease, 273, 274 Periprosthetic fractures, orthopedics, 445 Periprosthetic joint infections (PJIs), 444 Persistent GI bleeding, 995 Personal protective equipment (PPE), 251 Persons under investigation (PUI), 251 Pharyngocutaneous fistula, 254 Phosphate, 386 Physicians and cancer advice, 954, 955 Burton F. Dickey, MD, 954-955 chest X-ray, 954 description, 951 Marshall T. Morgan, MD, 951-952 Patrick J. Crocker, DO, 952-954 PET scans, 953, 954 RUQ pain, 951 PI3K inhibitors, 359 Pituitary adenomas, 230 Pituitary apoplexy, 963 clinical correlates of, 228 clinical features of, 228 Pituitary apoplexy score (PAS), 233 Pituitary dysfunction, pituitary tumor apoplexy, 228, 229 Pituitary macroadenoma, 225, 228 Pituitary tumor apoplexy anatomy and physiology, 227 clinical presentation, 227 headache, 227, 228 pituitary dysfunction, 228, 229 visual disturbance, 228 decision-making, algorithm for, 232 definition, 226 diagnostic imaging, 230 differential diagnosis, 230 emergency department management, 232, 233

incidence and predisposing factors, 229, 230 laboratory investigations, 230-232 neurosurgical management, 233 outcomes and follow up, 234 Plan-Do-Study-Act (PDSA), 24 Plastic stents, 334 Plateau wave, 962 Platinum-based therapy, 358 Pleural biopsy, 988 Pleural effusion, 308 chest tube size, 309 drainage, 308, 309 drainage, amount of, 309 Plicamycin, 384 Pneumatosis intestinalis, 361 Pneumocystis jirovecii pneumonia, 973 Pneumonia, 201 Pneumothorax, 307 cancer patients, clinical scenarios in, 307, 308 definitions, etiologies and diagnostic modalities, 306, 307 treatment, 308 Point-of-care ultrasound (POCUS), 307 See also Ultrasound Polyethylene glycol, 373 POMPE-C criteria, 320 Ponatinib, 273, 964 Poor care coordination, quality measures, 42, 43 Portal hypertension-related ascites, 336 Positive end-expiratory pressure (PEEP), 305 Positive predictive value (PPV), 27 Positron emission tomography (PET), superior vena cava syndrome, 296 Post-obstructive diuresis (POD), 410, 413, 414 Postrenal obstruction, 398 Powered air purifying respiratory (PAPR), 251 PPS-Exempt Cancer Hospitals Quality Reporting Program (PCHOR), 62 Premature coronary artery disease, 272, 273 Prescription drug abuse, 575 Prescription loss/replacement, 584 Pressure-volume curve, 305 Preventative medicine, 161 Primary cancer, 437 Primary health-care site, palliative social work, 111, 112 Primary hyperparathyroidism, 383 Probiotics, 362 Program in oncologic emergency medicine (POEM), DEM, 929-930 data repositories, 930 existing centers and departments, 931 goals, 930 interdisciplinary research hub, 930 manuscript advancement and development group, 930 program's mission, 930 PubLab, 930 scientific platforms, 931 Progressive multifocal leukoencephalopathy (PML), 208 Prolactin, 227 Prolactinoma, 233 Prolonged prerenal azotemia, 398 Proopiomelanocortin (POMC), 389 Proptosis, 264, 265 Propylthiouracil, 392 Prostate cancer, 418, 419 Prostatectomy catheter or urinary issues, 419 pelvic collections, 419

Prosthetic dislocation orthopedics, 445, 446 Protected health information (PHI), 32 Provider and patient awareness, ionizing radiation, 146 Pruritus, 334, 989 Psychoeducation, palliative social work, 114 Psychotherapeutic treatment approach, 583 Ptosis (upper eyelid, droopiness), 265, 266 Public health, 170-172 problems, 130 Pulmonary acute respiratory distress syndrome, 304 acute respiratory failure, 303 cancer survival, surveillance of, 303 chemotherapy-related pulmonary toxicity immune check point inhibitor related pulmonary toxicity, 310 lung toxicity, 311 consultation, 309 conventional chemotherapy related pulmonary toxicity, 311, 312 hemoptysis etiologies, 309, 310 massive hemoptysis, 310 non-thrombotic pulmonary embolism, 313 pleural effusion, 308 chest tube size, 309 drainage, 308, 309 drainage, amount of, 309 pneumothorax cancer patients, clinical scenarios in, 307, 308 definitions, etiologies and diagnostic modalities, 306, 307 treatment, 308 radiation-related pulmonary toxicity, 312 ventilator management, 304-306 Pulmonary edema syndrome, 311 Pulmonary embolism, 417 Pulmonary tuberculosis, 975 Purpuric eruptions, 464, 992 macular purpura, 465 palpable purpura, 465 retiform purpura, 465

Q

QT interval prolongation, 282 QT prolongation, 282 Quadrivalent vaccine, 154 Quality-adjusted life years (QALY), 26 Quality improvement, 24 Quality measures, 37 cancer care, current gaps, and measure development priorities, 62-63 caregiver burden, 41, 42 dedicated oncologic EDs, specific issues for, 42 delayed hospice referral and the hospice reimbursement model, 44, 45 Department of Health and Human Services, 66 driving practice change, role of, 67, 68 ED quality of care data, challenges in, 65 ED services, overutilization of, 39, 40 fragmented measure development, 64 gaps in, 64 health policy, history and current state of, 38 high costs at end of life, 40, 41 history of, 46, 62 immediate and after-hours outpatient care, limited availability of, 45 known quality issues, 39

Quality measures (cont.) late-stage cancers, 39 learning healthcare system, HIT, 67 limitations of, 64 National Quality Forum-endorsed measures, 47-61 No-Duty-to-Treat Principle and the Emergency Medical Treatment and Active Labor Act, 38, 39 oncologic emergency care health policy for, 66, 67 national quality measurement, vision for, 65, 66 state of national quality measurement for, 65 oncologic emergency care, episode of, 64 overcrowding, boarding and ambulance diversion, 40 palliative care, inadequate access to, 43, 44 patient dissatisfaction with emergency care, 41 poor care coordination, 42, 43 prognosis and treatment, unrealistic patient/caregiver expectations, 45,46 role of, 46 underutilized advance care planning, 43 upstream drivers, 42 without clear mechanism, 64, 65 Quality Payment Program, 147 Quinolone antibiotics, 363

R

Radiation, 141 acute toxicity, 663 after exposure, 664 amifostine, 665 gastrointestinal system, 664 granulocyte colony-stimulating factor, 665 hematopoietic systems, 664 hyperspectral optical imaging, 665 interleukin-1 alpha, 665 intracellular superoxide dismutase, 665 keratinocyte growth factor, 665 lymphocyte count, 664 mitigators, 665 normal tissue, 663 protectors, 665 sulfhydryl compounds, 665 total body doses, 664 total body exposure, 664 unintended exposure, 664 unintentional exposure, 665 colorectal cancer, 183 exposure, 663 late (delayed) effects, 668-675 bone marrow, 669 central and peripheral nervous system, 673-674 chemotherapy, 668 daily treatment dose, 668 endocrine organs, 675 gastrointestinal tract, 669-670 genitalia, 674 heart and peripheral vessel, 672, 673 hypothyroidism, 675 kidney, 670 liver, 670 lung, 671 QUANTEC, 668 reproductive organ, 674

skin, 668 therapy-associated hepatic injury, 670 normal tissue effects, 666 pediatrics, 675 unintentional exposure, 663 volume-modulated arc therapy, 675 Radiation enteritis, 352, 353 clinical presentation and initial assessment, 353 diagnostic evaluation, 353 treatment and operative intervention, 353 Radiation-induced hemorrhagic cystitis, 426 Radiation-induced liver disease (RILD), 670 Radiation-induced lung injury (RILI), 312 Radiation-induced painless thyroiditis, 391 Radiation induced valvular disease, 284 Radiation injury, 977 Radiation injury treatment network (RITN), 549 goals, 549 IND detonation, 549 medical staff, 549 Radiation proctitis, 330, 331, 426 Radiation therapy (RT), 273, 433 diastolic and restrictive cardiomyopathy, 279 gastrointestinal hemorrhage, 329 head and neck cancer, 254 manage acute effects, 666 daily treatment dose, 666 gastrointestinal tract, 667 hematopoietic system, 667 skin. 666 total radiation dose, 666 Radical cystectomy, 419 acute abdomen, 420 continent diversion problems, 421 dehvdration, 420 diarrhea, 420 fever, 419 hydronephrosis, 420 leg swelling, 420 stomal complications, 420 urinary tract infection, 420 Radioactivity, 1002 Radiological and nuclear incidents, 544 acute radiation syndrome (see Acute radiation syndrome (ARS)) biodosimetric techniques, 547 biodosimetry technique, 547 dose measurement, 544 edema and leukocyte infiltration, 551 improvised nuclear device, 544 internal contamination, 545 multiorgan failure, 551 NPP incidents, 545 nuclear weapon detonation, 544 pulmonary edema, 551 radioactivity measurement, 543 RDD, 544 RED, 544 Radiological dispersal device (RDD), 544 Radiological exposure device (RED), 544 Radiotherapy diarrhea, 359 superior vena cava syndrome, 298, 299 Rapid titration, 772

Rasburicase, 402 Rathke's cleft cyst, 226 Reactive oxygen species (ROS) production, 275 Recognize, Assist, Include, Support, and Engage (RAISE) Family Caregiver Act, 42 Recombinant granulocyte colony stimulating factor (G-CSF), 974 Rectal bleeding, 183, 351 Red eye, 263, 264 Refractory delirium, 996 Relative radiation level, 146 Relaxation techniques, 113 Renal cell carcinoma, 412, 417 Renal hypoperfusion, 398 Renal mass, 411 Renal mass related hematuria, 411 Renal replacement therapy (RRT), 399 Renin-angiotensin-aldosterone system (RAAS), 275 Renology, 397 acute kidney injury and anti-VEGF therapy, 400 cancer patients, 397-399 and conventional chemotherapy, 399, 400 and immunotherapy, 400 hematopoietic stem cell transplantation and, 401, 402 multiple myeloma and, 401 chronic kidney disease, in cancer patients, 400, 401 electrolyte abnormalities hypercalcemia, 404 hyperkalemia, 403, 404 hyponatremia, 402, 403 tumor lysis syndrome, 402 intrinsic renal disease, 398 postrenal obstruction, 398 prolonged prerenal azotemia, 398 renal hypoperfusion, 398 Reporting Hospital Quality Data for Annual Payment Update (RHQDAPU) program, 46 Respiratory depression, 979 Respiratory distress, 200 Restrictive cardiomyopathy anthracyclines and monoclonal antibodies, 279 radiation therapy, 279 Retiform purpura, 971 Retinal disease, 261 Retinal hemorrhage, 262 Retinal vasculature obstruction, 262 Retroperitoneal mass, 417 Rhegmatogenous retinal detachment, 261 5 Rights of Clinical Decision Support, 33 Rigid bronchoscope, 203 Rivaroxaban, 319, 322 Rome diagnostic criteria, constipation, 368

S

Saline laxatives, 372 Salivary fistula, 254 SARS-CoV-2, 97, 170 Screening, brief intervention and referral to treatment (SBIRT), 123, 134, 960 Screening, Treatment Initiation and Referral (STIR), 123 Secretagogues, 373 Seizures, 211, 213

Self-expanding metal stent (SEMS), 331 Sepsis, 431 Septic arthritis, orthopedics, 443, 444 Seroma, 253 Serum-ascites albumin gradient (SAAG), 336 Sexual minorities, 1001 Shared decision-making, 147 lung cancer screening, 171 Sheehan's syndrome, 226 Shorter survival, 998 Shortness of breath, 84-87 Sickle cell disease (SCD), 529 acute chest syndrome, 534 anti-coagulation role, 535 blood transfusion, 535 hypoventilation, 535 initial management, 535 acute coronary syndrome, 536 acute pain, 531, 538 aplastic crisis, 538 avascular necrosis, 537-538 blood transfusion, 539 cerebrovascular accidents, 536 chronic pain, 533 clinical and laboratory characteristics, 534 clinical manifestations, 531-532 clinical presentations, 531 emergency department management, 534 epidemiology, 531 fever and infection, 535 genotypes and characteristics, 531 HbSS genotype, 530 hematopoietic stem cell transplantation, 540 hepatobiliary, 537 hydroxyurea, 540 intravenous fluids, 539 ocular complications, 537 opioid crisis, 538 potential treatments, 531 priapism, 537 pulmonary embolism, 536 pulmonary hypertension, 535 randomized controlled trials, 540 renal complications, 537 splenic sequestration, 537 types of pain, 532 vaso-occlusive crisis (VOC), 533 Sinuses, 256 Sinusitis, 255 Six-chair unit, 15 Six Sigma, 24 Skin cancer, 191 burden of, 186 Small bowel intussusception, 344 Small bowel obstruction (SBO), 341, 342 clinical presentation and initial assessment, 342, 343 diagnostic workup, 343, 344 treatment and operative intervention, 344 Small cell LCs (SCLC), 163 SmokefreeTXT texting program, 124 Smoking, 121, 122 cessation, 126 Socioeconomic status (SES), 123

Specific serotonin reuptake inhibitors (SSRIs)., 570

Spinal cord compression, 39, 238, 240 clinical presentation history, 238, 239 imaging, 239, 240 physical examination, 239 epidemiology, 238 management, 240 medical therapy, 241 nursing efforts, 240, 241 pathophysiology, 238 prevention, 243 prognosis, 243 radiotherapy, 242 steroids, 242 surgical therapy, 241, 242 surveillance and diagnostic practices, 242 vielded significant, 238 Spinal instability neoplastic score (SINS) System, 241 Spinal tenderness, 239 Spiritual care personnel, 1000 Splenic rupture, 998 Spray cryotherapy, 329 Squamous cell carcinoma of the head and neck (SCCHN), 247 Squamous intraepithelial lesion (SIL), 152 Start osmotherapy, 963 Status epilepticus (SE) assessment, 214 causes, 213 definition and classification, 212, 213 management and prognosis, 215 Stem-cell transplant (SCT), 14, 95 Stents, 299, 300, 331 Stercoral colitis, 371 Stereotactic body radiation therapy (SBRT), 242 Stevens-johnson syndrome (SJS) SCORTEN severity of illness score for, 992 and toxic epidermal necrolysis (TEN), 461 body surface area, 462 clinical manifestations, 461, 462 diagnosis, 463 erythema multiforme, 462 pathophysiology, 463 SCORTEN severity, 463 systemic steroids, 464 treatment, 463 Stimulants, 979 laxatives, constipation, 373 Stochastic effects, 141 Stoma complications bleeding, 696 classification, 691-692 diarrhea, 692 health economics, 692 innovative strategies, 696 centralizing information, 697 ostomy telehealth, 697 prehabilitation, 696 PROMs, 697 overview, 691 parastomal hernia, 693, 694 peristomal dermatitis, 693 prolapse, 695, 696 retraction, 694 skin irritation, 693 stenosis, 694, 695 Stool softeners, constipation, 373

Stress cardiomyopathy, 277 diarrhea, 360 Stroke prevention trial in sickle cell anemia (STOP), 536 Subarachnoid bleeding, 233 Subconjunctival hemorrhage, 263 Substance Abuse and Mental Health Services Administration, 123 Substance use, 134 Substance use disorder (SUD), 60, 573 aberrant drug-taking behaviors and severity, 577 alcohol, 576 clinical management, 578 comorbid psychiatric disorders, 580 disease-related variables, 578 guidelines, 578 history, 579 multidisciplinary approach, 579 opioids, 574 pain and anxiety management, 574 prescription drug use, 575 prevalence, 574 pseudoaddiction, 577 realistic goals for therapy, 579 remote history of abuse, 578 risk assessment tools, 579 Subungual melanoma, 188 Suicide, 565, 971 airline industry, 565 antidepressants, 570 biopsychosocial model, 566 cancer patients, 565 ED in oncologic patients, 569 environmental factors, 568 future aspects, 570 genetic make-up anger and impulsivene behavior, 567 childhood trauma, 567 hypothalamus-pituitary-adrenal axis, 567 serotonin, 567 manage cancer patients, 568 oncologic clinic management, 569 precipitating factor chronic illnesses, 568 financial crisis/loss of job, 568 military veterans, 568 stress-diathesis model, 567, 568 refractory Hodgkin lymphoma, 566 risk factors to cancer patient, 568 theories, 566 traumatic effect, 565 Sun protection, 192 Sunbed (tanning bed) use, 192 Sunburn, 186, 192 Sunscreen, 192 Superficial keratopathy, 263 Superficial spreading melanoma, 187 Superior vena cava syndrome (SVCS), 596, 1002 anatomy, 292 benign, treatment of, 300 chemotherapy, 298 clinical features, 597 contrast-enhanced chest CT, 597 endovascular stenting, 597 etiology/epidemiology, 291, 292 histologic diagnosis, 296 intravascular therapy, 299, 300 malignant causes, treatment of, 298

palliative care, 300, 301 prognosis, 300 radiographic evaluation, 294-296 radiotherapy, 298, 299 recurrence, 300 signs/symptoms, 293, 294 supportive therapy, 297 symptoms, grading of, 296 thrombolytics, 300 treatment, 296 treatment, algorithm for, 297 Supportive counselling, palliative social work, 113 Surgery, diarrhea, 359 Surveillance, Epidemiology, and End Results (SEER) program, 34, 163 Symptomatic severe aortic stenosis, 284 Syndrome of inappropriate antidiuretic hormone secretion (SIADH), 997 Systemic fibrinolysis, 965 Systolic cardiomyopathies, 278 Systolic cardiomyopathy (CM), 277

Т

Tachycardia, 357 Tacrolimus, 387 Tamoxifen, 333, 371 Targeted therapy, diarrhea, 358, 359 Target-specific anticoagulants (TSA), 320 Tea and toast syndrome, 403 Teledermatology, melanoma, 189 Telehealth, 30, 31 history of, 30 oncologic emergencies and, 31 regulation, 31 technology, 30, 31 Telemedicine, 170 Tenosynovitis, orthopedics, 443 Testicular cancer, 415-417 Thalidomide, 275 Therapeutic bronchoscopy, 202 Thrombocytopenia, 262 Thrombogenic cancers, venous thromboembolism, 318 Thrombolysis, 300 Thrombolytics, superior vena cava syndrome, 300 Thrombosis anticoagulants in thrombocytopenic patients, 496 antineoplastic therapy, 496 central venous catheters, 495 epidemiology, 493 etiology, 494 LMWH vs DOAC, 494 myeloproliferative neoplasms, 494 PNH, 495 treatment, 494 Thrombotic microangiopathies (TM), 492 Thrombotic thrombocytopenic purpura (TTP), 492 Thyroid function tests, 372 Thyroid metastasis, 391 Thyroid storm, 392 Thyrotoxicosis, 377, 391, 392 Tissue factor (TF), 317 Tobacco control, 121 illnesses associated with tobacco use, 122-126 tobacco dependence treatment medications, 125 tobacco-related illness, diagnosis of, 123

tobacco use, diagnosis of, 121-123 Tobacco dependence treatment, 126 Tobacco-related illness, 121 diagnosis of, 123 Tobacco use, 121 diagnosis of, 121-123 Tolerance, 581 Total cost of care population-based PMPM Index, 55 Total hip replacements, 991 Toxic epidermal necrolysis, 992 Toxic erythema of chemotherapy (TEC), 455 angioedema, 455 clinical manifestations, 456 diagnosis, 456 pathophysiology/etiology, 456 treatment, 456 leukemia cutis, 455 Tracheoinnominate fistula, head and neck cancer, 252, 253 Tracheostomy, 250 Tramadol, 983 Tranexamic acid (TXA), 310 Transalar herniation, 975 Transcatheter aortic valve replacement (TAVR), 284 Transfusion-associated graft-versus-host disease (TA-GVHD), 549 Transition of care tools, 33, 34 Transthoracic echocardiogram (TTE), 280 Trauma, orthopedics, 437 Traumatic spinal cord injury, 238 Treatment-related pneumothorax, 307 Triage, 16 oncologic emergencies, 84 True epiphora (excessive tearing), 264 TSH deficiency, 230 Tumor fever, 94 Tumor lysis syndrome (TLS), 388, 402, 969 causes, 388, 389 diagnosis, 389 management, 389 symptoms, 389 Tumors, 84 12-step program, 584 Two-chair triage bay, 15 Typhlitis, 352, 361 Tyrosine kinase inhibitors (TKIs), 358 and VEGF signaling pathway inhibitors, 273

U

UCC, see Urgent Care Center (UCC) Ultra-low dose CT (ULDCT), 169 Ultrasound, 611 abdominal oncologic emergencies, 623-625 ascites and perforation, 623 colitis, 625 MBO, 623 B-mode images, 611 emergency cardiac, 614 apical four chamber, 614 cardiac tamponade, 614 consolidation, 621 curvilinear array transducer, 614 diffuse interstitial syndrome, 621 IVC measurement, 619 left ventricular systolic function, 616 lung artifacts, 621 parasternal views, 614

Ultrasound (cont.) pericardial effusions, 614 phased-array transducer, 614 pleural and parapneumonic effusions, 622 pleural disease, 622 pneumothorax, 623 right ventricular strain, 617 subxiphoid view, 614 thoracic and Pulmonary Oncologic Emergencies, 620 zones of lung, 620 gynecologic oncologic emergencies deep venous thrombosis, 626 obstructive urinary pathologies, 625 ovarian torsion, 625 pelvic pathologies, 625 imaging, 614 lymphadenopathy, 627-628 color flow, 628 hilar appearance, 628 margins, 628 size and shape, 628 structural changes, 628 superficial lymph nodes, 627 M-mode images, 611 pediatric neck masses, 629 RUSH protocol, 629 aortic slide view, 630 DVT, 630 ectopic pregnancy, 630 HI MAP ED approach, 629 IVC dilation, 629 Morison's pouch, 629 pulmonary views, 630 safety considerations, 614 time gain compensation, 614 transducer, 611 convex-shaped crystal arrays, 613 endocavitary transducers, 613 linear array transducers, 612 phased array transducers, 613 Ultraviolet (UV) radiation, 186 UVA, 186 UVB, 186 Underutilized advance care planning, quality measures, 43 Unilateral leg swelling, 996 Unilateral otalgia, 255 University of Texas MD Anderson Cancer Center, 15 Upper aerodigestive tract (UADT), 247 Upper gastrointestinal bleeding (UGIB), 328, 349 Upper urinary tract drainage, 415, 991 Upright abdominal series, 986 Upstream drivers, quality measures, 42 Ureteral masses, 411 Urethral pathology, 412, 413 Urgent Care Center (UCC), 16, 17 Urinary incontinence, 963 Urinary retention, 412-414 Urinary tract infection (UTI), 410 Urinary tract obstruction, gynecologic cancer, 431, 432 Urine drug testing (UDT), 585 Urology bladder masses, 411, 412 gross hematuria, 409-411 hematuria, 407 hydronephrosis, 414, 415 microhematuria, 407, 409

nephrectomy abdominal collections, 421 hematuria, 421 pulmonary symptoms, 421 penile cancer, 417, 418 postoperative issues, prostatectomy, 419 prostate cancer, 418, 419 radical cystectomy, 419 acute abdomen, 420 continent diversion problems, 421 dehvdration, 420 diarrhea, 420 fever, 419 hydronephrosis, 420 leg swelling, 420 stomal complications, 420 urinary tract infection, 420 renal mass, 411 retroperitoneal mass with venous involvement, 417 testicular cancer, 415-417 ureteral masses, 411 urethral pathology, 412, 413 urinary retention, 413, 414 Urostomy complications, 696 Uveitis, 264

V

Vaginal cuff cellulitis, 428 Vaginal cuff dehiscence, gynecologic cancer, 431 Vaginal packing, 427 Values, 1001 Valvular heart disease, 283 management of, 284 modern techniques, 284 pathophysiology, 283, 284 Varenicline, 124 Varicella-zoster virus (VZV) clinical presentation, 459 diagnosis, 459 pathophysiology/etiology, 459 treatment, 459 Vascular endothelial growth factor (VEGF), 271 Vascular endothelial growth factor inhibitors (VEGFIs), 274 Vasculitis, 273 Vaso-occlusive crisis, 533 Vasopressin secretion, 227 VEGFI-associated HTN, 274 Venous thromboembolism (VTE), 274, 275 advanced treatment, 323 for cancer patient, 318, 319 cancer therapies, 275 catheter-associated thrombosis, 322, 323 diagnosis, 275 criteria, 318 special considerations for, 319 as first manifestation of cancer, 324 follow up, 324 Hestia criteria, 320 incidental diagnosis and thrombophilia workup, 322 Khorana, risk tool of, 321 pathophysiological processes, 275 pathophysiology of, 317, 318 patients with absolute contraindications, 323 POMPE-C criteria, 320 prevention, 275, 276

prophylaxis, 321, 322 risk stratification and management, 319, 320 target-specific anticoagulants, randomized trials on, 321 thrombogenic cancers, 318 treatment, 276, 320, 321 treatment, discontinuation of, 323 worsened prognosis, 323 Ventricular arrhythmias, 282 Vertebroplasty, 242 Viral colitis, 362 Virchow's node, 415 Virus-like particles (VLPs), 153 Visual analogue scales (VAS), 766 Visual disturbance, pituitary tumor apoplexy, 228 Visual inspection with acetic acid (VIA), 156 Visualizing workflow, 24 Vitamin K, 183

Vitamin K antagonists (VKAs), 319, 320 Vitritis (inflammation of the vitreous gel), 266 VTE, *see* Venous thromboembolism (VTE)

\mathbf{W}

Warfarin-induced skin necrosis (WISN), 466 diagnosis, 467 pathophysiology, 467 treatment, 467 Water-soluble contrast agent (WSCA), 344 Watery diarrhea, 361 WHO analgesic ladder, 432, 768

Y

Y-tubing, 410